Eric Lichtfouse Marjolaine Hamelin Mireille Navarrete Philippe Debaeke *Editors* 







# Sustainable Agriculture Volume 2







Sustainable Agriculture Volume 2

#### **Previous Books**

#### Sustainable Agriculture

Lichtfouse E., Navarrete M., Debaeke P., Souchère V., Alberola C. (Eds.) Springer/EDP Sciences 2009. 919 p.

#### Genetics, Biofuels and Local Farming Systems

Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 7. Springer 2011. In press

#### Alternative Farming Systems, Biotechnology, Drought Stress and Ecological Fertilisation Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 6. Springer 2011. 354 p.

Biodiversity, Biofuels, Agroforestry and Conservation Agriculture

Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 5. Springer 2011. 391 p.

#### Genetic Engineering, Biofertilisation, Soil Quality and Organic Farming

Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 4. Springer 2010. 414 p.

#### Sociology, Organic Farming, Climate Change and Soil Science

Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 3. Springer 2010. 478 p.

#### Climate Change, Intercropping, Pest Control and Beneficial Microorganisms

Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 2. Springer 2010. 513 p.

#### **Organic Farming, Pest Control and Remediation of Soil Pollutants**

Lichtfouse E. (Ed.) Sustainable Agriculture Reviews. Volume 1. Springer 2009. 418 p.

#### Environmental Chemistry. Green Chemistry and Pollutants in Ecosystems

Lichtfouse E., Schwarzbauer J., Robert D. (Eds.) Springer 2005. 780 p.

#### Rédiger pour être publié! Conseils pratiques pour les scientifiques

Lichtfouse E. Springer 2009. 105 p. Eric Lichtfouse • Marjolaine Hamelin Mireille Navarrete • Philippe Debaeke Editors

# Sustainable Agriculture Volume 2



*Editors* Eric Lichtfouse INRA-CMSE-PME rue Sully 17 21000 Dijon France eric.lichtfouse@dijon.inra.fr

Marjolaine Hamelin French National Institute for Agriculture LBE Avenue des étangs 11100 Narbonne France marjolaine.hamelin@supagro.inra.fr Mireille Navarrete INRA-SAD Unite d'Écodéveloppement Avignon CX 09 France navarret@avignon.inra.fr

Philippe Debaeke INRA UMR AGIR PO Box 52627 31326 Toulouse CX France debaeke@toulouse.inra.fr

Cover design: Pictures were kindly provided by INRA colleagues. Copyright: Almonds: INRA/Christophe Maitre 2002; apples: INRA/François Laurent 2010; cherries: INRA/Michel Adrian 2010; oilseed rape field: INRA/Hervé Cochard 2010.

All contents have previously been published in the journal "Agronomy for Sustainable Development" published by EDP Sciences, France.

ISBN 978-94-007-0393-3 e-ISBN 978-94-007-0394-0 DOI 10.1007/978-94-007-0394-0 Springer Dordrecht Heidelberg London New York

Library of Congress Control Number: 2009926164

©Springer Science+Business Media B.V. 2011

No part of this work may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, microfilming, recording or otherwise, without written permission from the Publisher, with the exception of any material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

# Contents

#### Part I NOVEL CONCEPTS

<b>Emerging Agroscience</b> Eric Lichtfouse, Marjolaine Hamelin, Mireille Navarrete, Philippe Debaeke, and Agnès Henri	3
Ants and Sustainable Agriculture	15
Agroecology as a Science, a Movement and a Practice A. Wezel, S. Bellon, T. Doré, C. Francis, D. Vallod, and C. David	27
Adaptiveness to Enhance the Sustainability of Farming Systems Ika Darnhofer, Stéphane Bellon, Benoît Dedieu, and Rebecka Milestad	45
Economics of Biosecurity Across Levels of Decision-Making Jaakko Heikkilä	59
<b>Describing and Locating Cropping Systems on a Regional Scale</b> Delphine Leenhardt, Frédérique Angevin, Anne Biarnès, Nathalie Colbach, and Catherine Mignolet	85
Part II FOOD SECURITY	
Nutritional Quality and Safety of Organic Food            Denis Lairon	99
Minerals in Plant Food: Effect of Agricultural Practices and Role in Human Health	111
M.C. Martínez-Ballesta, R. Dominguez-Perles, D.A. Moreno, B. Muries, C. Alcaraz-López, E. Bastías, C. García-Viguera, and M. Carvajal	
<b>Fertiliser Trees for Sustainable Food Security in the Maize-Based</b> <b>Production Systems of East and Southern Africa</b> Festus K. Akinnifesi, O.C. Ajayi, G. Sileshi, P.W. Chirwa, and Jonas Chianu	129

Cereal Landraces for Sustainable Agriculture
Mineral Sources of Potassium for Plant Nutrition
Glandless Seed and Glanded Plant Research in Cotton
Micronutrient-Efficient Genotypes for Crop Yield and Nutritional Quality in Sustainable Agriculture
Multi-Criteria Decision Models for Management of Tropical Coastal Fisheries
Merlina N. Andalecio
Part III SOCIOLOGY AND ECONOMICS
<b>Farmer Responses to Climate Change and Sustainable Agriculture</b> 283 Aysha Fleming and Frank Vanclay
The Use of the Marasha Ard Plough for Conservation Agriculturein Northern Ethiopia295Jan Nyssen, Bram Govaerts, Tesfay Araya, Wim M. Cornelis, Hans Bauer,Mitiku Haile, Ken Sayre, and Jozef Deckers
<b>Biological Nitrogen Fixation and Socioeconomic Factors for Legume</b> <b>Production in Sub-Saharan Africa</b>
Conventionalisation of Organic Farming Practices: From Structural Criteria Towards an Assessment Based on Organic Principles
<b>Conservation Tillage in Turkish Dryland Research</b>
Part IV CLIMATE CHANGE
<b>Biofuels, Greenhouse Gases and Climate Change</b>
Agronomic and Physiological Performances of Different Speciesof Miscanthus, a Major Energy CropH.W. Zub and M. Brancourt-Hulmel

Changes in Atmospheric Chemistry and Crop Health
<b>Modelling Soil Carbon and Nitrogen Cycles During Land Use Change</b> 499 J. Batlle-Aguilar, A. Brovelli, A. Porporato, and D.A. Barry
Greenhouse Gases and Ammonia Emissions from Organic Mixed Crop-Dairy Systems: A Critical Review of Mitigation Options
Water Deficit and Nitrogen Nutrition of Crops557Victoria Gonzalez-Dugo, Jean-Louis Durand, and François Gastal
Validation of Biophysical Models: Issues and Methodologies
Cold Stress Tolerance Mechanisms in Plants
Part V ALTERNATIVE PEST CONTROL
<b>Defence Mechanisms of Brassicaceae: Implications for Plant-Insect</b> <b>Interactions and Potential for Integrated Pest Management</b>
Ionising Radiation and Area-Wide Management of Insect Peststo Promote Sustainable AgricultureMarc J.B. Vreysen and Alan S. Robinson
<b>Biodiversity and Pest Management in Orchard Systems</b>
Pathogenic and Beneficial Microorganisms in Soilless Cultures
Allelopathy in Compositae Plants
Part VI SOIL HEALTH
Assessing the Productivity Function of Soils
Long-Term Effects of Organic Amendments on Soil Fertility
<b>Tillage Management Effects on Pesticide Fate in Soils</b> 787Lionel Alletto, Yves Coquet, Pierre Benoit, Djilali Heddadj, and Enrique Barriuso

Sustainable Cow-Calf Operations and Water Quality
Gilbert C. Sigua
Biogeography of Soil Microbial Communities: A Review and a Description
of the Ongoing French National Initiative
Lionel Ranjard, Samuel Dequiedt, Claudy Jolivet, Nicolas P.A. Saby,
Jean Thioulouse, Jérome Harmand, Patrice Loisel, Alain Rapaport, Saliou
Fall, Pascal Simonet, Richard Joffre, Nicolas Chemidlin-Prévost Bouré,
Pierre-Alain Maron, Christophe Mougel, Manuel P. Martin, Benoît Toutain,
Dominique Arrouays, and Philippe Lemanceau
Part VII ALTERNATIVE FERTILISATION

Nitrogen Rhizodeposition of Legumes         869           Joëlle Fustec, Fabien Lesuffleur, Stéphanie Mahieu, and Jean-Bernard Cliquet         869
Models of Biological Nitrogen Fixation of Legumes
Arbuscular Mycorrhizal Networks: Process and Functions
Efficient N Management Using Winter Oilseed Rape
<b>Improving Nitrogen Fertilization in Rice by Site-Specific N Management</b> 943 Shaobing Peng, Roland J. Buresh, Jianliang Huang, Xuhua Zhong, Yingbin Zou, Jianchang Yang, Guanghuo Wang, Yuanying Liu, Ruifa Hu, Qiyuan Tang, Kehui Cui, Fusuo Zhang, and Achim Dobermann
<b>Solid–Liquid Separation of Animal Slurry in Theory and Practice</b>

# Contributors

**Majid Afyuni** Department of Soil Science, Isfahan University of Technology, 84154, Isfahan, Iran

**Ishita Ahuja** Department of Biology, Norwegian University of Science and Technology, Realfagbygget, NO-7491 Trondheim, Norway, atle.bones@bio.ntnu.no

**O.C. Ajayi** World Agroforestry Centre (ICRAF), Southern Africa Regional Programme, PO Box 30798, Lilongwe, Malawi

**T. Akar** Central Research Institute for Field Crops, PO Box 226, 06042 Ulus-Ankara, Turkey

**Festus K. Akinnifesi** World Agroforestry Centre (ICRAF), Southern Africa Regional Programme, PO Box 30798, Lilongwe, Malawi, fakinnifesi@africa-online.net

**C. Alcaraz-López** Plant Nutrition Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain

**Lionel Alletto** Université de Toulouse - École d'ingénieurs de Purpan, Agronomy Department, 75 voie du TOEC, BP 57611, 31076 Toulouse Cedex 3, France, lionel.alletto@purpan.fr

**Merlina N. Andalecio** Institute of Fisheries Policy and Development Studies, College of Fisheries and Ocean Sciences, University of the Philippines Visayas, Miag-ao, Iloilo 5023, Philippines, merlina\_andalecio@yahoo.com

**Frèdèrique Angevin** INRA, UAR1240 Eco-Innov, BP 01, 78850 Thiverval Grignon, France

**Tesfay Araya** Ghent University, Department of Soil Management, 9000 Gent, Belgium and

Mekelle University, Department of Crop and Horticultural Science, Mekelle, Ethiopia

**Dominique Arrouays** INRA Orléans - US 1106, Unité INFOSOL, avenue de la Pomme de Pin, BP 20619, Ardon, 45166 Olivet Cedex, France

**Muzaffer Avci** Central Research Institute Field Crops (CRIFC), Agronomy Department, PO Box 226, Ankara, Turkey, muzafavci@yahoo.com

**John A. Baddeley** Crop and Soil Systems Research Group, Scottish Agricultural College, Craibstone Estate, Aberdeen, AB21 9YA, UK

**Bruce C. Ball** Crop and Soil Systems Research Group, SAC, West Mains Road, Edinburgh EH9 3JG, UK

**J.P. Baresel** Technical University of Munich, Chair of Organic Farming, Alte Akademie 12, 85350 Freising, Germany

**Enrique Barriuso** UMR 1091 INRA/AgroParisTech Environment and Arable Crops, Institut National de la Recherche Agronomique/Institut National des Sciences et Industries du Vivant et de l'Environnement, 78850 Thiverval-Grignon, France

**D.A. Barry** Institute for Environmental Engineering, École Polytechnique Fédérale de Lausanne, Station 2, 1015, Lausanne, Switzerland, andrew.barry@epfl.ch

**Ruth Bartel-Kratochvil** BOKU – University of Natural Resources and Applied Life Sciences, Vienna, Feistmantelstr. 4, 1180 Vienna, Austria

**E. Bastías** Departamento de Producción Agrícola, Facultad de Ciencias Agronómicas, Universidad de Tarapacá, Casilla 6-D, Arica, Chile

**J. Batlle-Aguilar** School of the Environment, National Centre for Groundwater Research and Training (NCGRT), Flinders University, GPO Box 2100, Adelaide, SA-5001, Australia, jordi.batlleaguilar@flinders.edu.au

Hans Bauer Mekelle University, Department of Land Resources Management and Environmental Protection, Mekelle, Ethiopia and

K.U. Leuven, Department of Earth and Environmental Sciences, 3001 Heverlee, Belgium

**P.J. Bebeli** Department of Plant Breeding and Biometry, Agricultural University of Athens, Iera Odos 75, Athens 11855, Greece

**Gianni Bellocchi** Grassland Ecosystem Research Unit, French National Institute for Agricultural Research, 234 Avenue du Brézet, 63100 Clermont-Ferrand, France, gianni.bellocchi@clermont.inra.fr, giannibellocchi@yahoo.com

**Stéphane Bellon** UR 0767 Ecodéveloppement, INRA, 84914 Avignon Cedex 9, France

**Gero Benckiser** Department of Applied Microbiology, Justus-Liebig University, Giessen, Heinrich Buff-Ring 26-32, Germany, gero.benckiser@umwelt.uni-giessen.de

**Jürgen Bender** Institute of Biodiversity, Johann Heinrich von Thünen-Institute (vTI), Federal Research Institute for Rural Areas, Forestry and Fisheries, Bundesallee 50, 38116 Braunschweig, Germany, juergen.bender@vti.bund.de

**Pierre Benoit** UMR 1091 INRA/AgroParisTech Environment and Arable Crops, Institut National de la Recherche Agronomique/Institut National des Sciences et Industries du Vivant et de l'Environnement, 78850 Thiverval-Grignon, France **Cécile Bessou** INRA Environment and Agricultural Crop Research Unit, 78850 Thiverval-Grignon, France, cecile.bessou@cirad.fr

**E. Bettencourt** Genetic Resources, Ecophysiology and Plant Bredding Unit, Instituto Nacional dos Recursos Biológicos, I.P. (INRB, I.P.), Quinta do Marquês, 2784-505 Oeiras, Portugal

Anne Biarnès IRD, UMR1221 LISAH, 2 place Viala, 34060 Montpellier, France

**K.V. Bladenopoulos** NAGREF –Cereal Institute 57001, Thermi, Thessaloniki, Greece

**D. Blancard** UMR Santé Végétale 1065, INRA, ENITA de Bordeaux, Université de Bordeaux, 33175 Gradignan, France

Atle Magnar Bones Department of Biology, Norwegian University of Science and Technology, Realfagbygget, NO-7491 Trondheim, Norway, atle.bones@bio.ntnu.no

Nicolas Chemidlin-Prévost Bouré INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France and Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France

**Jean-Charles Bouvier** INRA, PSH Domaine St Paul, Agroparc, 84914 Avignon Cédex, France, jean-charles.bouvier@avignon.inra.fr

**M. Brancourt-Hulmel** INRA, USTL UMR 1281, 80203 Estrees-Mons, Peronne, France, brancour@mons.inra.fr

**A. Brovelli** Institute for Environmental Engineering, École Polytechnique Fédérale de Lausanne, Station 2, 1015, Lausanne, Switzerland

**Roland J. Buresh** Crop and Environmental Sciences Division, International Rice Research Institute, DAPO Box 7777, Metro Manila, Philippines

**Yingfan Cai** College of Bioinformation, Chongqing University of Posts and Telecommunications, Chongqing, 400065, China, caiyf3000@yahoo.com.cn

**M. Carvajal** Plant Nutrition Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain, mcarvaja@cebas.csic.es

**Shikha Chandel** Department of Botany, Panjab University, Chandigarh – 160014, India

**Rufus L. Chaney** USDA-Agricultural Research Service, Environmental Management and Byproduct Utilization Laboratory, Building 007, BARC-West, Beltsville, MD 20705, USA

Jonas N. Chianu TSBF-CIAT, c/o World Agroforestry Centre (ICRAF), UN Avenue, Gigiri, PO Box 30677, Nairobi, Kenya, j.chianu@cgiar.org

P.W. Chirwa Pretoria University, South Africa

**S.-U. Chon** EFARINET Co. Ltd., BI Center, Chosun University, Gwangju 501-759, South Korea, chonsu4100@yahoo.co.kr

**K.V. Christensen** Department of Chemical Engineering, Biotechnology and Environmental Technology, Faculty of Engineering, University of Southern Denmark, Niels Bohrs Allé 1, 5220 Odense, Denmark, kvc@kbm.sdu.dk

**M.L. Christensen** Department of Biotechnology, Chemistry and Environmental Engineering, Aalborg University, Sohngaardsholmsvej 57, 9000 Aalborg, Denmark, mlc@bio.aau.dk

**Jean-Bernard Cliquet** UMR INRA 950 Ecophysiologie Végétale Agronomie INRA/Université de Caen, 14032 Caen Cedex, France

Nathalie Colbach INRA, UMR1210 Biologie et Gestion des Adventices, 17 rue Sully, BP 86510, 21065 Dijon Cedex, France

**Yves Coquet** UMR 1091 INRA/AgroParisTech Environment and Arable Crops, Institut National de la Recherche Agronomique/Institut National des Sciences et Industries du Vivant et de l'Environnement, 78850 Thiverval-Grignon, France

Wim M. Cornelis Ghent University, Department of Soil Management, 9000 Gent, Belgium

**Kehui Cui** Crop Physiology and Production Center, MOA Key Laboratory of Huazhong Crop Physiology, Ecology and Production, Huazhong Agricultural University, Wuhan, Hubei 430070, China

**J.H. Czembor** Plant Breeding and Acclimatization Institute – IHAR Radzikow, 05-870 Blonie, Poland

Bahareh Daneshbakhsh Department of Soil Science, Isfahan University of Technology, 84154, Isfahan, Iran

**Ika Darnhofer** BOKU – University of Natural Resources and Applied Life Sciences, Vienna, Feistmantelstr. 4, 1180 Vienna, Austria and

Department of Economic and Social Sciences, University of Natural Resources and Applied Life Sciences Vienna, Feistmantelstr. 4, 1180 Vienna, Austria, ika.darnhofer@boku.ac.at

**C. David** ISARA, Department of Agroecosystems, Environment and Production, 23 rue Jean Baldassini, 69364 Lyon Cedex 07, France

Philippe Debaeke INRA, UMR AGIR, BP 52627, 31326 Toulouse Cedex, France

**Jean-François Debras** INRA, PSH Domaine St Paul, Agroparc, 84914 AVIGNON Cédex, France, jean-francois.debras@avignon.inra.fr

**Jozef Deckers** K.U. Leuven, Department of Earth and Environmental Sciences, 3001 Heverlee, Belgium

**Benoît Dedieu** UMR 1273 Metafort, INRA, Theix, 63122 Saint-Genès Champanelle, France

F. Déniel Université Européenne de Bretagne, France and

Université de Brest, EA3882 Laboratoire Universitaire de Biodiversité et Écologie Microbienne, IFR148 ScInBioS, ESMISAB, Technopôle Brest-Iroise, 29280 Plouzané, France

**Samuel Dequiedt** INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France and

Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France

Mariangela Diacono CRA- Research Unit for Cropping Systems in Dry Environments, Bari - Italy

Achim Dobermann Crop and Environmental Sciences Division, International Rice Research Institute, DAPO Box 7777, Metro Manila, Philippines

**R. Dominguez-Perles** Food Science and Technology Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain

**Marcello Donatelli** Agriculture Research Council, via di Corticella 133, 40128 Bologna, Italy

**T. Doré** AgroParisTech, UMR 211 INRA/AgroParisTech, BP 01, 78850 Thiverval-Grignon, France

**T.M.M. Dos Santos** ISOPlexis Banco de Germoplasma, BGR, CEM, Universidade da Madeira, 9000-390 Funchal, Portugal

**Jean-Louis Durand** INRA, Unité de Recherche Pluridisciplinaire sur les prairies et les plantes fourragères, BP 6, 86600 Lusignan, France

**Frank Eulenstein** Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany

**Saliou Fall** Équipe "Génomique Microbienne Environnementale" (Environmental Microbial Genomics Group), UMR CNRS 5005, Laboratoire Ampère, École Centrale de Lyon, 36 avenue Guy de Collongue, 69134 Ecully Cedex, France

**D.A. Fasoula** Agricultural Research Institute, P.O.Box 22016, 1516 Nicosia, Cyprus

Fabien Ferchaud INRA, US1158 Agro-Impact, 02 007 Laon-Mons, France

J.L. Fiorelli SOLPHY, La Grennery, 73670 Entremont-le-Vieux, France

**Aysha Fleming** Tasmanian Institute of Agricultural Research, University of Tasmania, Private Bag 54, Hobart TAS 7001, Australia, Aysha.Fleming@utas.edu.au

**C. Francis** University of Nebraska-Lincoln, Department of Agronomy and Horticulture, 279 Plant Science Hall, Lincoln, Nebraska 68583-0915, USA

**Joëlle Fustec** LUNAM University, Groupe ESA, UR Laboratoire d'Ecophysiologie Végétale et Agroécologie, 55 rue Rabelais, 49007 Angers Cedex 01, France, j.fustec@groupe-esa.com

Benoît Gabrielle INRA, US1158 Agro-Impact, 02 007 Laon-Mons, France

**C. García-Viguera** Food Science and Technology Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain

**Neera Garg** Department of Botany, Panjab University, Chandigarh – 160014, India, garg\_neera@yahoo.com

**François Gastal** INRA, Unité de Recherche Pluridisciplinaire sur les prairies et les plantes fourragères, BP 6, 86600 Lusignan, France

**Victoria Gonzalez-Dugo** IAS-CSIC, Instituto de Agricultura Sostenible, Consejo Superior de Investigaciones Científicas, Alameda del Obispo, s/n, 14004, Córdoba, Spain and

INRA, Unité de Recherche Pluridisciplinaire sur les prairies et les plantes fourragères, BP 6, 86600 Lusignan, France, victoria.gonzalez@ias.csic.es

**Bram Govaerts** International Maize and Wheat Improvement Centre (CIMMYT), México D.F. 06600, Mexico

**T. Graham Shepherd** BioAgriNomics Ltd., 6 Parata Street, Palmerston North 4410, New Zealand

**L. Guérin-Dubrana** UMR Santé Végétale 1065, INRA, ENITA de Bordeaux, Université de Bordeaux, 33175 Gradignan, France

**Mitiku Haile** Mekelle University, Department of Crop and Horticultural Science, Mekelle, Ethiopia

**Marjolaine Hamelin** INRA, UA1267, Agronomy for Sustainable Development, UR 50, Laboratoire de Biotechnologie de lŠEnvironnement, avenue des Étangs, 11100 Narbonne, France

**Jérome Harmand** INRA-INRIA MERE research project, UMR ASB, place Pierre Viala, 34060 Montpellier Cedex, France and

LBE-INRA, UR050, avenue des étangs, 11100 Narbonne, France

**Djilali Heddadj** Chambres d'agriculture de Bretagne, Recherche appliquée - Pôle agronomie, avenue du Général Borgnis Desbordes, BP 398, 56009 Vannes Cedex, France

**Jaakko Heikkilä** MTT Economic Research, Latokartanonkaari 9, 00790 Helsinki, Finland, jaakko.heikkila@mtt.fi

**Katharina Helming** Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany

**Agnès Henri** EDP Sciences, Parc dŠActivités de Courtaboeuf, BP 112, 17 avenue du Hoggar, 91944 Les Ulis Cedex A, France

**Maibritt Hjorth** Department of Biosystems Engineering, Faculty of Agricultural Sciences, Aarhus University, Blichers Allé 20, 8830 Tjele, Denmark

Amir Hossein Khoshgoftarmanesh Department of Soil Science, Isfahan University of Technology, 84154 Isfahan, Iran, amirhkhosh@cc.iut.ac.ir

**Ruifa Hu** Centre for Chinese Agricultural Policy, Institute of Geographical Sciences and Natural Resource Research, Chinese Academy of Sciences, Beijing 100101, China

**Jianliang Huang** Crop Physiology and Production Center, MOA Key Laboratory of Huazhong Crop Physiology, Ecology and Production, Huazhong Agricultural University, Wuhan, Hubei 430070, China

**Richard Joffre** UMR 5175 CNRS, Équipe DREAM - Centre d'Écologie Fonctionnelle et Évolutive, 1919 route de Mende, 34293 Montpellier Cedex 5, France

**Claudy Jolivet** INRA Orléans - US 1106, Unité INFOSOL, avenue de la Pomme de Pin, BP 20619, Ardon, 45166 Olivet Cedex, France

**Henning Kage** Institute of Crop Science and Plant Breeding, Christian-Albrechts-University, Hermann-Rodewald-Str. 9, 24118 Kiel, Germany

**A. Katsiotis** Department of Plant Breeding and Biometry, Agricultural University of Athens, Iera Odos 75, Athens 11855, Greece

**K. Koutis** Laboratory of Genetics and Plant Breeding, Faculty of Agriculture, Aristotle University of Thessaloniki, 54124 Thessaloniki, Greece

**M. Koutsika-Sotiriou** Department of Genetic Resources and Organic Plant Breeding, Agricultural Research Institute of the Hungarian Academy of Sciences, Brunszvik u. 2., Martonvasar, 2462, Hungary

**G. Kovacs** Department of Genetic Resources and Organic Plant Breeding, Agricultural Research Institute of the Hungarian Academy of Sciences, Brunszvik u. 2., Martonvasar, 2462, Hungary

Sudesh Kumar Yadav Biotechnology Division, Institute of Himalayan Bioresource Technology, CSIR, Palampur-176061 (HP), India, skyt@rediffmail.com; sudeshkumar@ihbt.res.in

**Denis Lairon** INRA, UMR 1260, Nutriments Lipidiques et Prévention des Maladies Métaboliques, U476, Univ. Aix-Marseille 1, Univ. Aix-Marseille 2, Faculté de Médecine, 13385 Marseille, France, denis.lairon@univmed.fr

**H. Larsson** Swedish University of Agricultural Sciences, Box 104, SE-23053 Alnarp, Sweden

**G. Le Floch** Université Européenne de Bretagne, France and

Université de Brest, EA3882 Laboratoire Universitaire de Biodiversité et Écologie Microbienne, IFR148 ScInBioS, ESMISAB, Technopôle Brest-Iroise, 29280 Plouzané, France **Delphine Leenhardt** INRA, UMR1248 – AGIR (Agrosystèmes et développement territorial), BP 52627, 31326 Castanet Tolosan Cedex, France, Delphine.Burger-Leenhardt@toulouse.inra.fr

**Philippe Lemanceau** INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France and

Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France

**Fabien Lesuffleur** UMR INRA 950 Ecophysiologie Végétale Agronomie INRA/Université de Caen, 14032 Caen Cedex, France

**Eric Lichtfouse** INRA, UA1267, Agronomy for Sustainable Development, INRA-CMSE-PME, 17 rue Sully, 21000 Dijon, France, eric.lichtfouse@dijon.inra.fr

**Thomas Lindenthal** BOKU – University of Natural Resources and Applied Life Sciences, Vienna, Feistmantelstr. 4, 1180 Vienna, Austria

**Jinggao Liu** Southern Plains Agricultural Research Center, United States Department of Agriculture, Texas, College Station, Texas, 77845, USA

Yanyan Liu College of Resources and Environmental Sciences, China Agricultural University, Beijing 100193, China

and

Crop and Soil Systems Research Group, Scottish Agricultural College, Craibstone Estate, Aberdeen, AB21 9YA, UK, yanyan.liu@cau.edu.cn

**Yuanying Liu** College of Resources and Environmental Sciences, Northeast Agricultural University, Harbin, Heilongjiang 150030, China

**Patrice Loisel** INRA-INRIA MERE research project, UMR ASB, place Pierre Viala, 34060 Montpellier Cedex, France

**F.S. Mairura** TSBF-CIAT, c/o World Agroforestry Centre (ICRAF), UN Avenue, Gigiri, PO Box 30677, Nairobi, Kenya

**Pierre-Alain Maron** INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France and

Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France

**Stéphanie Mahieu** LUNAM University, Groupe ESA, UR Laboratoire d'Ecophysiologie Végétale et Agroécologie, 55 rue Rabelais, 49007 Angers Cedex 01, France

**David A.C. Manning** School of Civil Engineering and Geosciences, Newcastle University, Newcastle upon Tyne, NE1 7RU UK, David.Manning@newcastle.ac.uk

**M.C. Martínez-Ballesta** Plant Nutrition Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain

**Manuel P. Martin** INRA Orléans - US 1106, Unité INFOSOL, avenue de la Pomme de Pin, BP 20619, Ardon, 45166 Olivet Cedex, France

Bruno Mary INRA, US1158 Agro-Impact, 02 007 Laon-Mons, France

Keith Matthews Macaulay Institute, Craigiebuckler AB15 8QH, Aberdeen, UK

**Rebecka Milestad** Dept. of Urban and Rural Studies, Swedish University of Agricultural Sciences, 75007 Uppsala, Sweden

**Wilfried Mirschel** Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany

**Catherine Mignolet** INRA, UR55 SAD-Mirecourt, 662 avenue Louis Buffet, 88500 Mirecourt, France

**Francesco Montemurro** CRA – Research Unit for the Study of Cropping Systems – Metaponto (MT), Italy, francesco.montemurro@entecra.it

**D.A. Moreno** Food Science and Technology Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain

**Christophe Mougel** INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France and

Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France

**Lothar Mueller** Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany, mueller@zalf.de

**B. Muries** Plant Nutrition Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100, Murcia, Spain

**Mireille Navarrete** INRA, UR 767, Écodéveloppement, 84914 Avignon Cedex 09, France

**C.J. Nelson** Department of Agronomy, University of Missouri, Columbia, MO 65211, USA

**A.C. Newton** SCRI, Invergowrie, Dundee DD2 5DA, Scotland, UK, adrian.newton@scri.ac.uk

**E.M. Nkonya** International Food Policy Research Institute (IFPRI), (202) 862-5600, 2033 K St NW, #400, Washington DC, USA

**S.M. Novak** INRA, Unité Expérimentale Fourrages et Environnement, F-86600 Lusignan, France, sandra.novak@lusignan.inra.fr

Jan Nyssen Department of Geography, Ghent University, 9000 Gent, Belgium, jan.nyssen@ugent.be

**Shaobing Peng** Crop and Environmental Sciences Division, International Rice Research Institute, DAPO Box 7777, Metro Manila, Philippines, s.peng@cgiar.org

**M.A.A. Pinheiro de Carvalho** ISOPlexis Banco de Germoplasma, BGR, CEM, Universidade da Madeira, 9000-390 Funchal, Portugal

**A. Porporato** Institute for Environmental Engineering, École Polytechnique Fédérale de Lausanne, Station 2, 1015, Lausanne, Switzerland and

Permanent address: Civil and Environmental Engineering Department, Duke University, Durham, NC 27708, USA

**Lionel Ranjard** INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France and

Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France, ranjard@dijon.inra.fr

Alain Rapaport INRA-INRIA MERE research project, UMR ASB, place Pierre Viala, 34060 Montpellier Cedex, France

**P. Rey** UMR Santé Végétale 1065, INRA, ENITA de Bordeaux, Université de Bordeaux, 33175 Gradignan, France, prey@bordeaux.inra.fr

Mike Rivington Macaulay Institute, Craigiebuckler AB15 8QH, Aberdeen, UK

Alan S. Robinson Entomology Unit, FAO/IAEA Agriculture and Biotechnology Laboratory, Joint FAO/IAEA Programme, Vienna, Austria

**Jutta Rogasik** Julius Kühn-Institut, Bundesforschungsinstitut für Kulturpflanzen (JKI), Institut für Pflanzenbau und Bodenkunde, Bundesallee 50, 38116 Braunschweig, Germany

Jens Rohloff Department of Biology, Norwegian University of Science and Technology, Realfagbygget, NO-7491 Trondheim, Norway

**D. Rubiales** Institute for Sustainable Agriculture, CSIC, Alameda del Obispo s/n, Apdo. 4084, 14080 Cordoba, Spain

J. Russell SCRI, Invergowrie, Dundee DD2 5DA, Scotland, UK

**Nicolas P.A. Saby** INRA Orléans - US 1106, Unité INFOSOL, avenue de la Pomme de Pin, BP 20619, Ardon, 45166 Olivet Cedex, France

**Benoît Sauphanor** INRA, PSH Domaine St Paul, Agroparc, 84914 AVIGNON Cédex 9, France, benoit.sauphanor@avignon.inra.fr

**Ken Sayre** International Maize and Wheat Improvement Centre (CIMMYT), México D.F. 06600, Mexico

**Uwe Schindler** Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany

**Rainer Schulin** Institutes of Terrestrial Ecology, ETH Zurich, Universitaetstr. 16, 8092 Zurich, Switzerland

**Klaus Sieling** Institute of Crop Science and Plant Breeding, Christian-Albrechts-University, Hermann-Rodewald-Str. 9, 24118 Kiel, Germany, sieling@pflanzenbau.uni-kiel.de **Gilbert C. Sigua** Research Soil Scientist, United States Department of Agriculture-Agricultural Research Service Subtropical Agricultural Research Station, Brooksville, FL, 34601, USA, gilbert.sigua@ars.usda.gov

**G. Sileshi** World Agroforestry Centre (ICRAF), Southern Africa Regional Programme, PO Box 30798, Lilongwe, Malawi

**Sylvaine Simon** INRA, UERI, Gotheron, 26320 Saint-Marcel-lès-Valence, France, sylvaine.simon@avignon.inra.fr

**Pascal Simonet** Équipe "Génomique Microbienne Environnementale" (Environmental Microbial Genomics Group), UMR CNRS 5005, Laboratoire Ampère, École Centrale de Lyon, 36 avenue Guy de Collongue, 69134 Ecully Cedex, France

**S.G. Sommer** Department of Biosystems Engineering, Faculty of Agricultural Sciences, Aarhus University, Blichers Allé 20, 8830 Tjele, Denmark, sgs@kbm.sdu.dk

**Qiyuan Tang** Crop Physiology, Ecology, and Production Center, Hunan Agricultural University, Changsha, Hunan 410128, China

**Jean Thioulouse** Université de Lyon, 69000 Lyon, France; Université Lyon 1, CNRS, UMR 5558, Laboratoire de Biométrie et Biologie Évolutive, 69622 Villeurbanne, France

**Benoît Toutain** INRA Orléans - US 1106, Unité INFOSOL, avenue de la Pomme de Pin, BP 20619, Ardon, 45166 Olivet Cedex, France

**J. Vallance** Université Européenne de Bretagne, France and

Université de Brest, EA3882 Laboratoire Universitaire de Biodiversité et Écologie Microbienne, IFR148 ScInBioS, ESMISAB, Technopôle Brest-Iroise, 29280 Plouzané, France

and

UMR Santé Végétale 1065, INRA, ENITA de Bordeaux, Université de Bordeaux, 33175 Gradignan, France

**D. Vallod** ISARA, Department of Agroecosystems, Environment and Production, 23 rue Jean Baldassini, 69364 Lyon Cedex 07, France

**Frank Vanclay** Faculty of Spatial Sciences, University of Groningen, PO Box 800, 9700 AV Groningen, The Netherlands, Frank.Vanclay@rug.nl

**M.C. Vaz Patto** Instituto de Tecnologia Química e Biológica, Apto. 127, 2781-901 Oeiras, Portugal

Marc J.B. Vreysen Insect Pest Control Laboratory, FAO/IAEA Agriculture and Biotechnology Laboratories, Joint FAO/IAEA Programme, Vienna, Austria, M.Vreysen@iaea.org

**Guanghuo Wang** College of Environmental and Natural Resources Sciences, Zhejiang University, Hangzhou, Zhejiang 310029, China **Christine A. Watson** Crop and Soil Systems Research Group, Scottish Agricultural College, Craibstone Estate, Aberdeen, AB21 9YA, UK

**Hans-Joachim Weigel** Institute of Biodiversity, Johann Heinrich von Thünen-Institute (vTI), Federal Research Institute for Rural Areas, Forestry and Fisheries, Bundesallee 50, 38116 Braunschweig, Germany

**A. Wezel** ISARA, Department of Agroecosystems, Environment and Production, 23 rue Jean Baldassini, 69364 Lyon Cedex 07, France, wezel@isara.fr

**Hubert Wiggering** Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany

Lianhai Wu Crop and Soil Systems Research Group, Scottish Agricultural College, Craibstone Estate, Aberdeen, AB21 9YA, UK and North Wyke Research, Okehampton, Devon EX20 2SB, UK

**Yongfang Xie** College of Bioinformation, Chongqing University of Posts and Telecommunications, Chongqing, 400065, China

**Jianchang Yang** Agronomy Department, Agricultural College, Yangzhou University, Yangzhou, Jiangsu 225009, China

**Fusuo Zhang** College of Resources and Environmental Sciences, China Agricultural University, Beijing 100094, China

**Xuhua Zhong** Rice Research Institute, Guangdong Academy of Agricultural Science, Guangzhou, Guangdong 510640, China

**Werner Zollitsch** BOKU – University of Natural Resources and Applied Life Sciences, Vienna, Feistmantelstr. 4, 1180 Vienna, Austria

**Yingbin Zou** Crop Physiology, Ecology, and Production Center, Hunan Agricultural University, Changsha, Hunan 410128, China

**H.W. Zub** INRA, USTL UMR 1281, 80203 Estrees-Mons, Peronne, France, brancour@mons.inra.fr

# Part I Novel Concepts

## **Emerging Agroscience**

Eric Lichtfouse, Marjolaine Hamelin, Mireille Navarrete, Philippe Debaeke, and Agnès Henri

#### Contents

1	Intro	oduction	4				
2	Most-Cited Articles 1999–2009						
3							
4	Mos	t-Downloaded Articles in 2009	8				
5	Revi	ew of Selected 2009 Articles	8				
	5.1	Sociology	8				
	5.2	Transgenic Crops	10				
	5.3	Climate Change	1(				
	5.4	Biodiversity	10				
	5.5	Alternative Farming Systems	11				
6	Cone	clusion	11				
Re	eferen	ces	11				

E. Lichtfouse (S)

Abstract Climate change and the recent financial crisis clearly show that humans have entered the anthropocene, an unprecedented era of fast and possibly dangerous changes. Unprecedented changes call for unprecedented thinking. Indeed, agricultural research has been for too long driven solely by the need for higher yields using classical agrosciences, whatever the adverse ecological effects. Agricultural research needs the input of other sciences such as ecological, economic, social and political sciences. These social sciences emerged in agricultural research a few decades ago, but there are currently no precise trends and data on the speed of emergence of specific topics. Therefore, here we report: (1) an analysis of the emergence of topics in the journal Agronomy for Sustainable Development, and (2) a review of selected articles published in 2009. First, to analyse topic emergence we studied three data sets: most-cited articles from 1999 to 2009, topic hits in article text from 1999 to 2009, and most-downloaded articles in 2009. We found the following major points. Most-cited articles show that transgenic plants and biofuels are clearly emerging topics from 2007, whereas soil carbon and climate change are the major mainstream topics of the last 10 years. Topic hits analysis allows one to rank topics by mean emergence date, e.g. 2008.3 for 'genetically modified' and 2005.3 for 'irrigation'. Accordingly, the 10 most emerging topics over 1999-2009 are biofuels, genetically modified, conservation agriculture, urban agriculture, sociology, organic farming, carbon sequestration, phytoremediation, mulch and biodiversity. Analysis of most-downloaded articles in 2009 shows the predominance of topics such as carbon, climate, biodiversity, biofuels, pollutants, beneficial microbes, transgenic plants and organic farming. Second,

INRA, UA1267, Agronomy for Sustainable Development, INRA-CMSE-PME, 17 rue Sully, 21000 Dijon, France e-mail: eric.lichtfouse@dijon.inra.fr

<sup>E. Lichtfouse et al. (eds.), Sustainable Agriculture Volume 2, DOI 10.1007/978-94-007-0394-0\_1,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Lichtfouse et al., Agron. Sustain. Dev. 30 (2010) 1–10. DOI: 10.1051/agro:2009055</sup> 

we reviewed selected articles published in 2009 with emphasis on emerging topics. We find that sociology is clearly bringing novel and unexpected findings to designing sustainable agriculture. Transgenic crops are highly innovative but show many unknowns that need to be carefully studied using various disciplines. Climate change has many scientifically proven effects on terrestrial ecosystems and agriculture. Here, soil carbon loss should be of particular attention because it rules the long-term fate of many factors such as atmospheric  $CO_2$ , erosion, and water and nutrient supply. Biodiversity loss due to industrial monocropping is leading scientists to disclose alternative, more diverse cropping systems that optimise biodiversity, pest control and yield.

**Keywords** Agriculture • Climate change • Biofuel • Transgenic plants • Biodiversity • Sociology • Organic farming • Conservation agriculture • Urban agriculture • Industrial agriculture • Carbon sequestration • Soil carbon • Crop rotation • No tillage • Beneficial microbes • Citation analysis • Topic emergence analysis Most-cited articles • Most-downloaded articles

#### 1 Introduction

Climate change and the recent financial crisis clearly show that humans have entered an unprecedented era of fast and possibly dangerous changes. This era is the anthropocene, a term that was coined in 2000 by the Nobel Prize-winning atmospheric chemist Paul Crutzen to point out that human activities now have a global impact on climate and ecosystems. Crutzen has explained, "I was at a conference where someone said something about the Holocene. I suddenly thought this was wrong. The world has changed too much. So I said: 'No, we are in the anthropocene'. I just made up the word on the spur of the moment. Everyone was shocked. But it seems to have stuck".

Unprecedented changes call for unprecedented adaptation. Unprecedented adaptation calls for unprecedented thinking. For instance, a major issue is that agricultural research has been for too long driven solely by the need for *higher yields* using monoculture, whatever the adverse ecological effects, such as food and drinking water pollution, biodiversity loss, and pest resistance. Mainstream goals such as higher yields should be challenged and rethought to take into account other factors. Those factors should not be solely defined by classical agrosciences, e.g. plant and soil sciences, but should also include all other sciences that really rule agriculture; for instance, ecological, economic, social and political sciences (De Bon et al., 2009; Wezel et al., 2009; Fleming and Vanclay, 2009; Lamine and Bellon, 2009; Lichtfouse et al., 2009a, b; Veldkamp et al., 2009). In other words, agronomy should not be reduced to a science that improves crop yields but should answer all society's issues because agriculture is both the foundation and the future of society. Agronomists should rethink the role of agriculture in our society. For instance, studies of farming systems should also include food systems (Gliessman, 2007).

Previous works have attempted to define the core issues of the industrial agricultural society: technology without wisdom; and the tragedy of the global commons: soil, water and air (Lal, 2009a, b); artificialisation and painkiller solutions; climate change and outdated society structures; and society dependence and sustainable agriculture (Lichtfouse, 2009a-c). In the book Sustainable Agriculture we have gathered 53 review articles that cover major advances in agrosciences (Lichtfouse et al., 2009a). Four recent books also report major contributions in emerging agrosciences (Lichtfouse et al., 2009d-g). Here, we analyse recent topical trends in the journal Agronomy for Sustainable Development. Topical trends are assessed using three indicators: (1) most-cited articles 1999-2009, (2) topic hits on the journal website over 1999-2009, and (3) most-downloaded articles in 2009. We then briefly review selected journal articles published in 2009.

#### 2 Most-Cited Articles 1999–2009

A first means to assess topics in the journal is to look at most-cited articles in the *Journal Citation Reports*. Table 1 shows yearly top-cited articles in the journal *Agronomy for Sustainable Development* from 1999 to 2009. The results indicate that major recent topics are transgenic plants, agroindicators, alternative crop management, beneficial microbes, and topics related to climate change such as biofuels and soil carbon. Whereas

**Table 1** Top 3 yearly most cited articles in the journal Agronomy for Sustainable Development. Number of citing articles, named cites, are from ISI-thompson on October 22, 2009. The top 5 highest cites are printed in bold. Full references are given in the reference list

Most-C	Cited Arti	cles	
Year	Cites	Topics	First author - Title
2008	5	Transgenic plants	BONNY. Genetically modified glyphosate-tolerant soybean in the USA
	5	Transgenic plants	DEVOS. Feasibility of isolation perimeters for genetically modified maize
	4	Agroindicators, N, pesticides	BOCKSTALLER. Agri-environmental indicators to assess cropping and farming
2007	16	Transgenic plants	DEVOS. Implementing isolation perimeters around genetically modified
	11	Alternative fertilisation, beneficial microbes	GARG. Symbiotic nitrogen fixation in legume nodules: process and signaling.
	10	Biofuels, climate, carbon	HILL. Environmental costs and benefits of transportation biofuel production
2006	17	Alternative management, soil, carbon	BERNOUX. Cropping systems, carbon sequestration and erosion in Brazil.
	17	Alternative fertilisation, soil, carbon	HACHICHA. Compost of poultry manure and olive mill wastes as an alternative
	8	Pollutants, food	GROVA. Effect of oral exposure to polycyclic aromatic hydrocarbons
2005	19	Alternative management, soil, pesticides	LACAS. Using grassed strips to limit pesticide transfer to surface water.
	18	Agroindicators, biodiversity	CLERGUE. Biodiversity: function and assessment in agricultural areas.
	13	Transgenic plants	COLBACH. Spatial aspects of gene flow between rapeseed varieties and volunteers.
2004	25	Alternative management, biodiversity	MARRIOTT. Long-term impacts of extensification of grassland management
	21	Alternative management, soil, erosion	LE BISSONNAIS. Grass strip effects on runoff and soil loss.
	17	Soil, carbon, compost	AMIR. Elemental analysis, FTIR and C-13-NMR of humic acids from sewage
2003	98	Soil, carbon, roots	NGUYEN. Rhizodeposition of organic C by plants: mechanisms and controls.
	26	Soil, carbon, fungi	CASARIN. Quantification of oxalate ions and protons released by ectomycorrhizal
	23	Pollutants, toxicity	ROUT. Effect of metal toxicity on plant growth and metabolism: I. Zinc.
2002	146	Alternative management, soil, carbon	SIX. Soil organic matter, biota and aggregation in temperate and tropical soils
	50	Agroindicators, climate, soil	BRISSON. STICS: a generic model for simulating crops and their water
	33	Climate, greenhouse	REICHRATH. Using CFD to model the internal climate of greenhouses
2001	51	Climate, carbon, salt stress	DORAI. Influence of electric conductivity management on greenhouse tomato
	35	Alternative control	EHRET. Disinfestation of recirculating nutrient solutions in greenhouse horticulture.
	32	Carbon, growth	GUICHARD. Tomato fruit quality in relation to water and carbon fluxes.
2000	72	Climate, canopy	WEISS. Investigation of a model inversion technique to estimate canopy
	52	Alternative control, intercropping	FINCKH. Cereal variety and species mixtures in practice
1000	37	Beneficial microbes	BOSSIS. The taxonomy of Pseudomonas fluorescens and Pseudomonas putida
1999	73	Climate, remote sensing	CEROVIC. Ultraviolet-induced fluorescence for plant monitoring
	44	Climate, canopy	GOUNNIER. ADEL-maize: an L-system based model for the integration of growth
	40	Plant architecture, model	GODIN. Exploration of a plant architecture database with the AMAPmod

We also studied topics by counting hits over 1999–2009 using the search engine of the journal website (Table 2). As topics refer to queries found in the whole article text, topics do not necessarily represent article topic. We ranked topics by decreasing order of emergence using the mean emergence date calculated by averaging dates weighted by hits. Figures 1 and 2 show examples of the evolution of topic hits with time. Four evolution types were identified: (1) A *plateau*; for

transgenic plants and biofuels are clearly emerging topics from 2007, soil carbon and climate are strikingly apparent as major topics through the whole 1999–2009 period. Soil carbon and climate are also topics of the 4 most-cited articles from 1999 to 2009. Other mostcited topics include agroindicators; alternative crop management, pest control and fertilisation; biodiversity; and pollutants and pesticides. We conclude that transgenic plants and biofuels are clearly emerging topics, whereas soil carbon and climate are the major mainstream topics of the last 10 years.

**Table 2** Mean emergence date (MED) of topics in articles from the journal *Agronomy for Sustainable Development*. Topics were searched in all article text, and thus do not necessarily reflect the article topic. Topics that showed highest hits in 2009 are printed in bold. Hits in percent refer to the number of articles containing topics measured using the journal website search engine on October 22–23, 2009, versus total yearly published ar-

ticles. Mean emergence date was calculated as the average of years weighed by hits:  $\sum$ (Hits.year)/ $\sum$ Hits, thus allowing to sort topics by order of emergence in the journal. Most recent topics have thus most recent date (see Fig. 1). For queries having several words, e.g. genetically modified, we used double quotes operators ("...") to retrieve only answers from words appearing together. AD: Anno Domini

	MED	2009	2008	2007	2006	2005	2004	2003	2002	2001	2000	1999
TOPICS	AD	%	%	%	%	%	%	%	%	%	%	%
Biofuels	2008.29	5.48	3.39	2.27	0	0	0	0	0	0	0	0
Genetically modified	2008.25	5.48	5.08	2.27	0	0	0	0	0	0	0	0
Conservation	2007.54	13.70	1.69	0	9.09	0	1.92	0	0	0	0	0
agriculture												
Urban agriculture	2007.22	2.74	1.69	0	0	0	1.92	0	0	0	0	0
Sociology	2007.11	10.96	0	2.27	0	1.59	3.85	1.27	0	0	0	0
Organic farming	2007.05	38.36	20.34	22.73	9.09	7.94	0	3.80	5.13	1.43	2.50	0
Carbon sequestration	2006.92	16.44	11.86	6.82	9.09	4.76	3.85	0	3.85	0	0	0
Phytoremediation	2006.87	1.37	6.78	2.27	0	3.17	0	1.27	0	0	0	0
Mulch	2006.62	21.92	28.81	15.91	15.15	3.17	9.62	1.27	5.13	1.43	1.25	1.79
Biodiversity	2006.55	54.79	25.42	25	12.12	12.7	7.69	5.06	6.41	5.71	2.50	5.36
Climate change	2006.54	31.51	18.64	9.09	9.09	6.35	9.62	2.53	5.13	2.86	1.25	1.79
Integrated pest	2006.50	9.59	10.17	11.36	6.06	4.76	0	3.8	0	1.43	0	1.79
management												
Allelopathy	2006.41	5.48	6.78	4.55	3.03	1.59	0	2.53	0	2.86	0	0
Soil erosion	2006.19	21.92	16.95	20.45	15.15	6.35	9.62	1.27	7.69	0	2.50	3.57
Crop rotation	2006.17	49.32	30.51	31.82	18.18	12.70	11.54	6.33	12.82	10	2.50	5.36
Transgenic	2006.15	13.70	13.56	13.64	0	4.76	1.92	6.33	0	0	1.25	5.36
Grass strips	2005.92	5.48	3.39	2.27	6.06	1.59	1.92	0	0	1.43	2.50	0
Biocontrol	2005.87	5.48	8.47	4.55	12.12	3.17	3.85	6.33	0	2.86	0	0
Cover crops	2005.76	17.81	16.95	15.91	6.06	9.52	7.69	3.80	14.1	2.86	1.25	1.79
Biological control	2005.70	10.96	10.17	18.18	12.12	7.94	7.69	7.59	0	7.14	2.50	0
No tillage	2005.59	42.47	35.59	34.09	18.18	20.63	28.85	18.99	20.51	5.71	3.75	8.93
Weed control	2005.35	27.40	27.12	18.18	24.24	20.63	19.23	13.92	8.97	14.29	2.50	7.14
Intercropping	2005.34	16.44	13.56	18.18	9.09	4.76	5.77	3.80	8.97	4.29	3.75	7.14
Irrigation	2005.21	39.73	42.37	43.18	39.39	39.68	51.92	37.97	32.05	28.57	0	0
Biological nitrogen	2005.00	6.85	6.78	11.36	0	0	1.92	3.80	1.28	14.29	0	0
fixation												
Agroforestry	2004.88	10.96	6.78	6.82	3.03	1.59	1.92	0	3.85	2.86	3.75	7.14
Decision support	2004.74	8.22	8.47	6.82	3.03	0	13.46	3.80	6.41	5.71	1.25	3.57
systems												
Precision agriculture	2004.06	4.11	3.39	2.27	0	1.59	9.62	1.27	6.41	2.86	0	3.57
Drought stress	2004.01	12.33	5.08	13.64	0	6.35	11.54	10.13	7.69	10	2.5	12.5



**Fig. 1** Emergence of topics in article text from the journal *Agronomy for Sustainable Development*. Note the shift to the right of most emerging topics in the journal. MED refers to mean emergence date (see Table 3 caption). Hits in percent refer to the number of articles containing topics measured using the journal website search engine (see Table 3 caption)



**Fig. 2** Evolution of the topic Agroforestry in article text from the journal *Agronomy for Sustainable Development*. Note the decrease until 2003 followed by an increase, suggesting a renewed interest. Hits in percent refer to the number of articles containing topics measured u sing the journal website search engine (see Table 3 caption)

instance, 'irrigation' data shows a plateau with a mean emergence date of 2005.2. (2) A *regular increase* such as that for 'crop rotation' since 2003. 'Crop rotation' has thus a younger mean emergence date of 2006.2. (3) A *sharp increase* such as that for 'organic farming' after 2004. 'Organic farming' has thus an even younger mean emergence date of 2007.1. 'Genetically modified' exhibits the youngest mean emergence date of 2008.3. (4) A *dive-rise* such as that for 'agroforestry', showing a decrease from 1999 to 2003 followed by an increase from 2003 to 2009, suggesting a renewed interest in this topic.

Table 2 shows that according to the mean emergence date the 10 most emerging topics are biofuels, genetically modified, conservation agriculture, urban agriculture, sociology, organic farming, carbon sequestration, phytoremediation, mulch and biodiversity. This finding confirms biofuels and transgenic plants as emerging topics from citation data (Table 1). The predominance of soil carbon and climate is also apparent in most emerging topics.

Urban agriculture, ranking 3 in emergence, and sociology, ranking 4, are of special interest because those topics represent a clear change of thinking. Urban agriculture is challenging the common belief that crops should be cultivated in rural areas. Here, the idea of producing food close to consumers to decrease transportation pollution and costs is clearly elegant and apposite to fight climate change. The emergence of sociology can be explained both by increased interest from agronomists and by the shift in journal topics from 2004 (Lichtfouse et al., 2004). Table 2 also shows printed in bold the 5 top topics according to hits in 2009. Here, biodiversity is found in 55% of articles, crop rotation 49%, no tillage 43%, irrigation 40% and organic farming 38%. These data from one single year cannot be interpreted in terms of evolution, but they probably represent the major concerns of authors in 2009.

To conclude, the 10 most emerging topics according to mean emergence date are biofuels, genetically modified organisms, conservation agriculture, urban agriculture, sociology, organic farming, carbon sequestration, phytoremediation, mulch and biodiversity. Those topics can be roughly classified into two streams of research aiming at a more sustainable agriculture. First, an analytical stream that develops technological innovations in plant science such as transgenic plants and biofuels. Second, a systemic stream that develops innovative farming practices such as organic farming and urban agriculture. Concepts of the systemic stream are given by Hill and MacRae (1996), Vandermeer et al. (1998), Papy (2001), Dalgaard et al. (2003), Lichtfouse et al. (2009b), and references therein.

#### 4 Most-Downloaded Articles in 2009

Topics of interest for readers of the journal can be evaluated by topics of the most-downloaded articles on the journal website (Table 3). We observe three major categories of topics: (1) topics related to climate change, e.g. biofuels, drought and salt stress, biodiversity, and carbon sequestration in soils; (2) topics related to alternative management, e.g. agroindicators, fertilisation, beneficial microbes, intercropping and organic farming, and (3) topics related to food security, e.g. pollutants, alternative fertilisation and control, and organic farming. The predominance of topics such as carbon, climate, biodiversity, biofuels, pollutants, beneficial microbes, transgenic plants and organic farming agrees with our previous results.

#### 5 Review of Selected 2009 Articles

Here, we review selected articles published in 2009 in the journal *Agronomy for Sustainable Development*. The fast emergence of new disciplines such as social and economic sciences in agronomic research is underlined by Wezel et al. (2009), Lamine and Bellon (2009), and Lichtfouse et al. (2009b). Wezel et al. (2009) reconstruct the historical rise of agroecology following the decline of industrial agriculture. It is noteworthy that these authors observe that the meaning of agroecology changes from one country to another. For instance, in Germany agroecology is mainly a science, whereas in France and in Brazil agroecology is also a green movement that is run by citizens.

#### 5.1 Sociology

Sociology emergence in the journal Agronomy for Sustainable Development is highlighted well by Veldkamp et al. (2009), who designed an innovative Dutch project that takes into account the opinion of all stakeholders. The authors develop their ideas on the alternative principle that sustainable development requires a better balance of the triple P values – people, planet and prosperity. Another tantalising example of bridging sociology and agronomy is given by Fleming and Vanclay (2009), who analyse the impact of climate change on farmer discourses. The core issue is nicely underlined by a farmer interview: "What's sustainable? You've got to look at our world as we know it. We're not in a sustainable position at the moment. That's why I say 'what's sustainable?' - I don't know". The authors found that farmers have four main discourses that guide their decision: money, earth, human responsibility and questioning. Disclosing which discourses are at work in a specific farming system is clearly needed because, to put it simply, farmers will not follow scientists' advice if they are not convinced that this advice agrees with their discourses. De Bon et al. (2009) show the social benefits of urban agriculture in developing countries. Lamine and Bellon (2009) review the conversion from intensive to organic farming using viewpoints from both agronomists and social scientists. They found that most publications report conversion effects and motivations, whereas few publications study transitions and trajectories. To conclude, the use of sociological tools to study farming systems is bringing unexpected findings to designing sustainable agriculture.

Most-Dov	wnloaded Art	ticles 2009*	
Year	PDF	Topics	First author - Title
2009	1975	Climate, drought stress	FAROOQ. Plant drought stress: effects, mechanisms and management.
2007	1421	Alternative fertilisation, beneficial microbes	KHAN. Role of phosphate-solubilizing microorganisms
2009	1260	Agroecology	LICHTFOUSE. Agronomy for sustainable agriculture.
2003	970	Pollutants, toxicity	ROUT. Effect of metal toxicity on plant growth
2006	931	Alternative fertilisation, beneficial microbes	HAFEEZ. Plant growth-promoting bacteria as biofertilizer
2009	786	Alternative management, intercropping	MALEZIEUX. Mixing plant species in cropping systems
2007	647	Biofuel, climate, carbon	HILL. Environmental costs and benefits of transportation biofuel
2001	641	Alternative fertilisation, beneficial microbes	IGUAL. Phosphate-solubilizing bacteria as inoculants
2009	639	Soil, food security	LAL. Soils and food sufficiency.
2009	629	Alternative management, soil	LAL. Laws of sustainable soil management.
2009	628	Climate, biodiversity	FEEHAN. Climate change in Europe. 1. Impact on terrestrial
2009	609	Climate, carbon, sequestration	BENBI. A 25-year record of carbon sequestration
2001	522	Climate, carbon, salt stress	DORAI. Influence of electric conductivity management
2009	497	Transgenic plants	DEVOS. Coexistence of genetically modified
E-first	495	Organic farming, food security	LAIRON. Nutritional quality and safety of organic food.
2009	493	Organic farming, sociology	LAMINE. Conversion to organic farming
2009	492	Agroindicators	BOCKSTALLER. Comparison of methods to assess the sustainability
2009	473	Climate, agriculture	LAVALLE. Climate change in Europe. 3. Impact on agriculture
2005	446	Climate, drought stress	SAMARAH. Effects of drought stress on growth and yield of barley.
2003	437	Climate, salt stress	BEN KHALED. Effet du stress salin en milieu hydroponique
2009	436	Climate, soil	JONES. Climate change in Europe. 2. Impact on soil.
2008	427	Alternative control	DORDAS. Role of nutrients in controlling plant diseases
2009	423	Agroecology	WEZEL. Agroecology as a science, a movement and a practice.
2007	422	Alternative fertilisation, beneficial microbes	GARG. Symbiotic nitrogen fixation in legume nodules
2009	411	Alternative fertilisation, intercropping	ZUO. Iron and zinc biofortification strategies in dicot plants
2003	392	Soil, carbon, roots	NGUYEN. Rhizodeposition of organic C by plants
E-first	391	Alternative fertilisation, food security	SPIERTZ. Nitrogen, sustainable agriculture and food security.
E-first	366	Alternative management, urban agriculture	DE BON. Sustainable urban agriculture in developing countries.
2003	364	Pollutants, beneficial microbes	JONER. Phytoremediation of organic pollutants using mycorrhizal
2003	341	Beneficial microbes, methods	GAMALERO. Methods for studying root colonization

Table 3 Top 30 most downloaded articles from the website of the journal Agronomy for Sustainable Development

\* Counts from January 1, 2009 to October 27, 2009. Year refers to year of publication. PDF refers to number of articles downloaded. PDF: portable document format. Full references are given in the reference list. E-first refers to articles online published but not yet issue published; those articles will be published in 2010

#### 5.2 Transgenic Crops

Transgenic crops is typically an emerging topic that has escaped the science sphere to be now a major social, economical and political issue. As a consequence, transgenic crops is therefore a well-suited research topic for agronomists that use sociology and economics. Like all major scientific breakthroughs, e.g. nitro-glycerine and nuclear energy, there are many arguments in favour of genetically modified organisms and many arguments against their use. Devos et al. (2009) analyse policies ruling the coexistence of transgenic and non-transgenic maize in European nations. They found that current isolation distances are excessive, difficult to implement, and not economically viable. Very interestingly, they conclude that 'other scientific issues must be at play'. Such an unknown may indeed be disclosed by the discourse approach of Fleming and Vanclay (2009).

Graef (2009) reviews possible adverse effects of introducing transgenic oilseed rape into Europe. He found that possible adverse effects to be monitored are persistence and spread of herbicide-tolerant oilseed rape, transfer of tolerance to wild relatives, development of herbicide tolerance of weeds, decrease in biodiversity, herbicide pollution, and adverse impact on field organisms and biogeochemical cycles. Hart et al. (2009) demonstrate for the first time the persistence of transgenic crop DNA residues within a soil food web. They found the transgene for glyphosate tolerance in soil arthropods, nematodes and earthworms from a transgenic corn field. The potential of transgene flow from transgenic crops' wild relatives is addressed by Loureiro et al. (2009), who found that hybrids of wheat and its wild relative Aegilops biuncialis are formed easily with 9-75% hybridisation rates. To conclude, transgenic crops are highly innovative but show both benefits and drawbacks that need to be carefully studied using various disciplines.

#### 5.3 Climate Change

Climate change effects in Europe are reviewed in three reprints from an European report. First, Feehan et al. (2009) analyse major effects on terrestrial ecosystems and biodiversity. Key trends include northward and uphill shift of plants, birds and mammals; and earlier seasonal events such as flowering, bird nesting and frog spawning. Second, Jones et al. (2009) report major effects of climate change on soils. Key observations include a decrease in soil C over the last 25 years, a higher risk of erosion and projected increases in CO<sub>2</sub> release in the atmosphere. It should be noted that there is much less evidence from soil studies because soil research is usually more difficult and needs more time - and funds - because most soil changes occur very slowly. Here, the major issue is that once adverse changes have occurred, e.g. soil carbon depletion, they cannot be healed fast. Therefore, management options that favour carbon sequestration and soil preservation should be applied (Doumbia et al., 2009; Benbi and Brar, 2009; Hazarika et al., 2009; Pleguezuelo et al., 2009; Tuttobene et al., 2009). Principles for sustainable soil management are given by Lal (2009c, d).

Third, Lavalle et al. (2009) review major effects of climate change on agriculture and forestry. Key trends include the shortening of the growing season in the south with higher risk of frost damage; flowering and maturity of crops occurring now about 2-3 weeks earlier; a higher yield variability due to extreme climate events such as the 2003 summer heat and the 2007 spring drought; a high increase of 50-70% in the water demand in Mediterranean areas; a faster forest growth; and a higher risk of forest fires. Tingem et al. (2009) simulate future crop yield in response to climate change in Cameroon. They found that developing later-maturing cultivars could greatly increase yields of maize, sorghum and bambara groundnut. Farooq et al. (2009) review effects of drought stress on plants. They also propose several solutions to counteract drought stress. To conclude, there are many scientific proofs of the effects of climate change on terrestrial ecosystems. Adverse, long-term effects such as soil carbon loss and erosion should be paid particular attention by agronomists.

#### 5.4 Biodiversity

Biodiversity loss due to adverse effects of industrial agriculture is a major threat to sustainable agriculture. A well-known example is the decrease in bees. Indeed, bees and other insects carry pollen and thus are essential to the reproduction of some crops such as blueberries. Higher biodiversity is also a means to control crop diseases because pathogens and natural antagonists are better balanced. De Cauwer and Reheul (2009) investigate the impact of grassland management on plant biodiversity and invasive species. They found that abundance of non-leguminous dicots decreases with higher intensity use and N supply. They identify several land uses that suppress invasive species and optimise plant biodiversity. Weed diversity and density is also investigated by Izquierdo et al. (2009) and Koocheki et al. (2009). Pelosi et al. (2009) studied earthworm biodiversity in conventional, organic and living-mulch cropping systems. They found that the Shannon-Wiener and equitability indexes were higher in the livingmulch system. To conclude, agronomists are designing alternative cropping systems that optimise biodiversity, pest control and yield.

#### 5.5 Alternative Farming Systems

Alternative management systems involving mixing plants, e.g. intercropping and agroforestry, are reviewed by Malézieux et al. (2009). They found that potential benefits are higher overall productivity, better pest control and better ecological services. Koocheki et al. (2009) analysed the effect of different cropping systems and various crop rotations on weeds. To reduce the weed seed bank they propose rotations that include crops with different life cycles such as winter wheat-maize and winter wheat-sugar beet. To design and evaluate innovative cropping systems on a medium- and long-term basis, new methods are required combining simulation and field experimentation. Here, Debaeke et al. (2009) describe an iterative and rule-based approach to setting up cropping systems in response to a drastic reduction of water, nitrogen and pesticide inputs.

#### 6 Conclusion

Citation analysis shows that transgenic plants and biofuels are clearly emerging topics, whereas soil carbon and climate change are the major mainstream topics of the last 10 years. The 10 most emerging topics according to mean emergence date are biofuels, genetically modified, conservation agriculture, urban agriculture, sociology, organic farming, carbon sequestration, phytoremediation, mulch and biodiversity. In 2009, the highest hits were found for biodiversity (55%), crop rotation (49%), no tillage (43%) and organic farming (38%). Analysis of most-downloaded articles in 2009 shows the predominance of topics such as carbon, climate, biodiversity, biofuels, pollutants, beneficial microbes, transgenic plants and organic farming. We find that sociology is clearly bringing novel and unexpected findings to designing sustainable agriculture. Transgenic crops are highly innovative but show many unknowns that need to be carefully studied using various disciplines. Climate change has many scientifically proven effects on terrestrial ecosystems and agriculture. Here, soil carbon loss should be of particular attention because it rules the long-term fate of many factors such as atmospheric CO<sub>2</sub>, erosion, and water and nutrient supply. Biodiversity loss due to industrial agriculture is leading scientists to disclose alternative, more diverse cropping systems that optimise biodiversity, pest control and yield.

#### References

- Amir S., Hafidi M., Merlina G., Hamdi H., Revel. J.C. (2004) Elemental analysis, FTIR and C-13-NMR of humic acids from sewage sludge composting, Agronomie 24, 13–18.
- Ben Khaled L., Gomez A.M., Honrubia M., Oihabi. A. (2003) Effects of salt stress in hydroponic media on clover plants inoculated with Rhizobium, Agronomie 23, 553–560.
- Benbi D.K., Brar. J.S. (2009) A 25-year record of carbon sequestration and soil properties in intensive agriculture, Agron. Sustain. Dev. 29, 257–265.
- Bernoux M., Cerri C.C., Cerri C.E.P., Neto M.S., Metay A., Perrin A.-S., Scopel E., Razafimbelo T., Blavet D., Piccolo M.d.C., Pavei M., Milne E. (2006) Cropping systems, carbon sequestration and erosion in Brazil, a review, Agron. Sustain. Dev. 26, 1–8.
- Bockstaller C., Guichard L., Keichinger O., Girardin P., Galan M.B., Gaillard. G. (2009) Comparison of methods to assess the sustainability of agricultural systems. A review, Agron. Sustain. Dev. 29, 223–235.
- Bockstaller C., Guichard L., Makowski D., Aveline A., Girardin P., Plantureux. S. (2008) Agri-environmental indicators to assess cropping and farming systems. A review, Agron. Sustain. Dev. 28, 139–149.
- Bonny S. (2008) Genetically modified glyphosate-tolerant soybean in the USA: adoption factors, impacts and prospects. A review, Agron. Sustain. Dev. 28, 21–32.
- Bossis E., Lemanceau P., Latour X., Gardan L. (2000) The taxonomy of *Pseudomonas fluorescens* and *Pseudomonas putida*: current status and need for revision, Agronomie 20, 51–63.

- Brisson N., Ruget F., Gate P., Lorgeau J., Nicoullaud B., Tayot X., Plenet D., Jeuffroy M.H., Bouthier A., Ripoche D., Mary B., Justes E. (2002) STICS: a generic model for simulating crops and their water and nitrogen balances. II. Model validation for wheat and maize, Agronomie 22, 69–92.
- Casarin V., Plassard C., Souche G., Arvieu. J.C. (2003) Quantification of oxalate ions and protons released by ectomycorrhizal fungi in rhizosphere soil, Agronomie 23, 461–469.
- Cerovic Z.G., Samson G., Morales F., Tremblay N., Moya. I. (1999) Ultraviolet-induced fluorescence for plant monitoring: present state and prospects, Agronomie 19, 543–578.
- Clergue B., Amiaud B., Pervanchon F., Lasserre-Joulin F., Plantureux S. (2005) Biodiversity: function and assessment in agricultural areas. A review, Agron. Sustain. Dev. 25, 1–15.
- Colbach N., Molinari N., Meynard J.M., Messean A. (2005) Spatial aspects of gene flow between rapeseed varieties and volunteers, Agron. Sustain. Dev. 25, 355–368.
- Dalgaard T., Hutchings N.J., Porter J.R. (2003) Agroecology, scaling and interdisciplinarity, Agr. Ecosyst. Environ. 100, 39–51.
- De Bon H., Parrot L., Moustier P. (2009) Sustainable urban agriculture in developing countries. A review, Agron. Sustain. Dev., DOI: 10.1051/agro:2008062.
- De Cauwer B., Reheul D. (2009) Impact of land use on vegetation composition, diversity and potentially invasive, nitrophilous clonal species in a wetland region in Flanders, Agron. Sustain. Dev. 29, 277–285.
- Debaeke P., Munier-Jolain N., Bertrand M., Guichard L., Nolot J.-M., Faloya V., Saulas P. (2009) Iterative design and evaluation of rule-based cropping systems: methodology and case studies. A review, Agron. Sustain. Dev. 29, 73–86.
- Devos Y., Demont M., Dillen K., Reheul D., Kaiser M., Sanvido O. (2009) Coexistence of genetically modified (GM) and non-GM crops in the European Union. A review, Agron. Sustain. Dev. 29, 11–30.
- Devos Y., Reheul D., Thas O., De Clercq E.M., Cougnon M., Cordemans K. (2007) Implementing isolation perimeters around genetically modified maize fields, Agron. Sustain. Dev. 27, 155–165.
- Devos Y., Thas O., Cougnon M., De Clercq E.M., Cordemans K., Reheul D. (2008) Feasibility of isolation perimeters for genetically modified maize, Agron. Sustain. Dev. 28, 195–206.
- Dorai M., Papadopoulos A.P., Gosselin A. (2001) Influence of electric conductivity management on greenhouse tomato yield and fruit quality, Agronomie 21, 367–383.
- Dordas C. (2008) Role of nutrients in controlling plant diseases in sustainable agriculture. A review, Agron. Sustain. Dev. 28, 33–46.
- Doumbia M., Jarju A., Sene M., Traore K., Yost R., Kablan R., Brannan K., Berthe A., Yamoah C., Querido A., Traore P.C.S., Ballo A. (2009) Sequestration of organic carbon in West African soils by Amenagement en Courbes de Niveau, Agron. Sustain. Dev. 29, 267–275.
- Ehret D.L., Alsanius B., Wohanka W., Menzies J.G., Utkhede R. (2001) Disinfestation of recirculating nutrient solutions in greenhouse horticulture, Agronomie 21, 323–339.

- Farooq M., Wahid A., Kobayashi N., Fujita D., Basra. S.M.A. (2009) Plant drought stress: effects, mechanisms and management, Agron. Sustain. Dev. 29, 185–212.
- Feehan J., Harley M., Minnen J. (2009) Climate change in Europe. 1. Impact on terrestrial ecosystems and biodiversity. A review, Agron. Sustain. Dev. 29, 409–421.
- Finckh M.R., Gacek E.S., Goyeau H., Lannou C., Merz U., Mundt C.C., Munk L., Nadziak J., Newton A.C., Vallavieille-Pope C., Wolfe M.S. (2000) Cereal variety and species mixtures in practice, with emphasis on disease resistance, Agronomie 20, 813–837.
- Fleming A., Vanclay F. (2009) Farmer responses to climate change and sustainable agriculture. A review, Agron. Sustain. Dev., DOI: 10.1051/agro/2009028.
- Fournier C., Andrieu B. (1999) ADEL-maize: an L-system based model for the integration of growth processes from the organ to the canopy. Application to regulation of morphogenesis by light availability, Agronomie 19, 313–327.
- Gamalero E., Lingua G., Berta G., Lemanceau P. (2003) Methods for studying root colonization by introduced beneficial bacteria, Agronomie 23, 407–418.
- Garg N., Geetanjali (2007) Symbiotic nitrogen fixation in legume nodules: process and signaling. A review, Agron. Sustain. Dev. 27, 59–68.
- Gliessman S.R. (2007) Agroecology, the ecology of sustainable food systems, Second edition, CRC Press, Taylor & Francis, ISBN: 978-0-8493-2845-9, 384 p.
- Godin C., Guedon Y., Costes E. (1999) Exploration of a plant architecture database with the AMAPmod software illustrated on an apple tree hybrid family, Agronomie 19, 163–184.
- Graef F. (2009) Agro-environmental effects due to altered cultivation practices with genetically modified herbicide-tolerant oilseed rape and implications for monitoring. A review, Agron. Sustain. Dev. 29, 31–42.
- Grova N., Rychen G., Monteau F., Le Bizec B., Feidt C. (2006) Effect of oral exposure to polycyclic aromatic hydrocarbons on goat's milk contamination, Agron. Sustain. Dev. 26, 195–199.
- Guichard S., Bertin N., Leonardi C., Gary C. (2001) Tomato fruit quality in relation to water and carbon fluxes, Agronomie 21, 385–392.
- Hachicha S., Chtourou M., Medhioub K., Ammar. E. (2006) Compost of poultry manure and olive mill wastes as an alternative fertilizer, Agron. Sustain. Dev. 26, 135–142.
- Hafeez F.Y., Yasmin S., Ariani D., Mehboob R., Zafar Y., Malik K.A. (2006) Plant growth-promoting bacteria as biofertilizer, Agron. Sustain. Dev. 26, 143–150.
- Hart M.M., Powell J.R., Gulden R.H., Levy-Booth D.J., Dunfield K.E., Pauls K.P., Swanton C.J., Klironomos J.N., Trevors J.T. (2009) Detection of transgenic cp4 epsps genes in the soil food web, Agron. Sustain. Dev. 29, 497–501.
- Hazarika S., Parkinson R., Bol R., Dixon L., Russell P., Donovan S., Allen D. (2009) Effect of tillage system and straw management on organic matter dynamics, Agron. Sustain. Dev. 29, 525–533.
- Hill J. (2007) Environmental costs and benefits of transportation biofuel production from food- and lignocellulose-based energy crops. A review, Agron. Sustain. Dev. 27, 1–12.

- Hill S.B., MacRae R.J. (1996) Conceptual framework for the transition from conventional to sustainable agriculture, J. Sustain. Agr. 7, 81–87.
- Igual J.M., Valverde A., Cervantes E., Velazquez E. (2001) Phosphate-solubilizing bacteria as inoculants for agriculture: use of updated molecular techniques in their study, Agronomie 21, 561–568.
- Izquierdo J., Blanco-Moreno J.M., Chamorro L., Gonzalez-Andujar J.L., Sans F.X. (2009) Spatial distribution of weed diversity within a cereal field, Agron. Sustain. Dev. 29, 491–496.
- Joner E.J., Leyval C. (2003) Phytoremediation of organic pollutants using mycorrhizal plants: a new aspect of rhizosphere interactions, Agronomie 23, 495–502.
- Jones A., Stolbovoy V., Rusco E., Gentile A.R., Gardi C., Marechal B., Montanarella L. (2009) Climate change in Europe. 2. Impact on soil. A review, Agron. Sustain. Dev. 29, 423–432.
- Khan M.S., Zaidi A., Wani P.A. (2007) Role of phosphatesolubilizing microorganisms in sustainable agriculture - A review, Agron. Sustain. Dev. 27, 29–43.
- Koocheki A., Nassiri M., Alimoradi L., Ghorbani R. (2009) Effect of cropping systems and crop rotations on weeds, Agron. Sustain. Dev. 29, 401–408.
- Lacas J.-G., Voltz M., Gouy V., Carluer N., Gril J.-J. (2005) Using grassed strips to limit pesticide transfer to surface water: a review, Agron. Sustain. Dev. 25, 253–266.
- Lairon D. (2009) Nutritional quality and safety of organic food. A review, Agron. Sustain. Dev., DOI: 10.1051agro2009019.
- Lal R. (2009a) Technology Without Wisdom, in: Lichtfouse E. (Ed.), Organic Farming, Pest Control and Remediation of Soil Pollutants. Sustainable Agriculture Reviews, Vol. 1, Springer, pp. 11–14.
- Lal R. (2009b) Tragedy of the Global Commons: Soil, Water and Air, in: Lichtfouse E. (Ed.), Climate Change, Intercropping, Pest Control and Beneficial Microorganisms. Sustainable Agriculture Reviews, Vol. 2, Springer, pp. 9–11.
- Lal R. (2009c) Laws of sustainable soil management, Agron. Sustain. Dev. 29, 7–10.
- Lal R. (2009d) Soils and food sufficiency. A review, Agron. Sustain. Dev. 29, 113–133.
- Lamine C., Bellon S. (2009) Conversion to organic farming: a multidimensional research object at the crossroads of agricultural and social sciences. A review, Agron. Sustain. Dev. 29, 97–112.
- Lavalle C., Micale F., Houston T.D., Camia A., Hiederer R., Lazar C., Conte C., Amatulli G., Genovese G. (2009) Climate change in Europe. 3. Impact on agriculture and forestry. A review, Agron. Sustain. Dev. 29, 433–446.
- Le Bissonnais Y., Lecomte V., Cerdan O. (2004) Grass strip effects on runoff and soil loss, Agronomie 24, 129–136.
- Lichtfouse E. (2009a) Sustainable Agriculture as a Central Science to Solve Global Society Issues, in: Lichtfouse E. (Ed.), Organic Farming, Pest Control and Remediation of Soil Pollutants. Sustainable Agriculture Reviews, Vol. 1, Springer, pp. 1–3.
- Lichtfouse E. (2009b) Climate Change, Society Issues and Sustainable Agriculture, in: Lichtfouse E. (Ed.), Climate Change, Intercropping, Pest Control and Beneficial Microorganisms. Sustainable Agriculture Reviews, Vol. 2, Springer, pp. 1–7.

- Lichtfouse E. (2009c) Society issues, painkiller solutions, dependence and sustainable agriculture, in: Lichtfouse E. (Ed.), Sociology, Organic Farming, Climate Change and Soil Science. Sustainable Agriculture Reviews, Vol. 3, Springer, pp. 1–17.
- Lichtfouse E. (2009d) Organic Farming, Pest Control and Remediation of Soil Pollutants. Sustainable Agriculture Reviews, Vol. 1, Springer, 418 p.
- Lichtfouse E. (2009e) Climate Change, Intercropping, Pest Control and Beneficial Microorganisms. Sustainable Agriculture Reviews, Vol. 2, Springer, 514 p.
- Lichtfouse E. (2009f) Sociology, Organic Farming, Climate Change and Soil Science. Sustainable Agriculture Reviews, Vol. 3, Springer, 478 p., ISBN: 978-90-481-3332-1, in press.
- Lichtfouse E. (2009g) Genetic engineering, biofertilisation, soil quality and organic farming. Sustainable Agriculture Reviews, Vol. 4, Springer, Approx. 400 p., ISBN: 978-90-481-8740-9, in press.
- Lichtfouse E., Habib R., Meynard J.-M., Papy F. (2004) Agronomy for Sustainable Development, Agronomie 24, 445.
- Lichtfouse E., Navarrete M., Debaeke P., Souchère V., Alberola C. (2009a) Sustainable Agriculture, Springer - EDPS, ISBN: 978-90-481-2665-1, 919 p.
- Lichtfouse E., Navarrete M., Debaeke P., Souchère V., Alberola C., Menassieu J. (2009b) Agronomy for sustainable agriculture. A review, Agron. Sustain. Dev. 29, 1–6.
- Loureiro I., Escorial C., Garcia-Baudin J.M., Chueca C. (2009) Hybridization, fertility and herbicide resistance of hybrids between wheat and Aegilops biuncialis, Agron. Sustain. Dev. 29, 237–245.
- Malézieux E., Crozat Y., Dupraz C., Laurans M., Makowski D., Ozier-Lafontaine H., Rapidel B., de Tourdonnet S., Valantin-Morison M. (2009) Mixing plant species in cropping systems: concepts, tools and models. A review, Agron. Sustain. Dev. 29, 43–62.
- Marriott C.A., Fothergill M., Jeangros B., Scotton M., Louault F. (2004) Long-term impacts of extensification of grassland management on biodiversity and productivity in upland areas. A review, Agronomie 24, 447–462.
- Nguyen C. (2003) Rhizodeposition of organic C by plants: mechanisms and controls, Agronomie 23, 375–396.
- Papy F. (2001) L'interdépendance des systèmes de culture dans l'exploitation agricole, in: Malézieux E., Trébuil G., Jaeger M. (Eds.), Modélisation des agro-éco-systèmes et aide à la décision, Editions Quae, Cirad-Inra, coll. Repères, pp. 51– 74, ISBN: 978-2-8761-4474-3.
- Pelosi C., Bertrand M., Roger-Estrade J. (2009) Earthworm community in conventional, organic and direct seeding with living mulch cropping systems, Agron. Sustain. Dev. 29, 287–295.
- Pleguezuelo C.R.R., Zuazo V.H.D., Raya A.M., Martinez J.R.F., Rodrigues B.C. (2009) High reduction of erosion and nutrient losses by decreasing harvest intensity of lavender grown on slopes, Agron. Sustain. Dev. 29, 363–370.
- Reichrath S., Davies T.W. (2002) Using CFD to model the internal climate of greenhouses: past, present and future, Agronomie 22, 3–19.
- Rout G.R., Das P. (2003) Effect of metal toxicity on plant growth and metabolism: I. Zinc, Agronomie 23, 3–11.
- Samarah N.H. (2005) Effects of drought stress on growth and yield of barley, Agron. Sustain. Dev. 25, 145–149.

- Six J., Feller C., Denef K., Ogle S.M., de Moraes J.C., Albrecht A. (2002) Soil organic matter, biota and aggregation in temperate and tropical soils - Effects of no-tillage, Agronomie 22, 755–775.
- Spiertz J.H.J. (2009) Nitrogen, sustainable agriculture and food security. A review, Agron. Sustain. Dev., DOI: 10.1051agro2008064.
- Tingem M., Rivington M., Bellocchi G. (2009) Adaptation assessments for crop production in response to climate change in Cameroon, Agron. Sustain. Dev. 29, 247–256.
- Tuttobene R., Avola G., Gresta F., Abbate V. (2009) Industrial orange waste as organic fertilizer in durum wheat, Agron. Sustain. Dev. 29, 557–563.
- Vandermeer J., Noordwijk M., Anderson J., Chin O., Perfecto I. (1998) Global change and multi-species agroecosystems: Concepts and issues, Agr. Ecosyst. Environ. 67, 1–22.
- Veldkamp A., Van Altvorst A.C., Eweg R., Jacobsen E., Van Kleef A., Van Latesteijn H., Mager S., Mommaas H., Smeets P., Spaans L., Van Trijp J.C.M. (2009) Triggering transitions towards sustainable development of the Dutch agricultural sector: TransForum's approach, Agron. Sustain. Dev. 29, 87–96.
- Weiss M., Baret F., Myneni R.B., Pragnere A., Knyazikhin Y. (2000) Investigation of a model inversion technique to estimate canopy biophysical variables from spectral and directional reflectance data, Agronomie 20, 3–22.
- Wezel A., Bellon S., Doré T., Francis C., Vallod D., David C. (2009) Agroecology as a science, a movement and a practice. A review, Agron. Sustain. Dev. 29, 503–515.
- Zuo Y., Zhang F. (2009) Iron and zinc biofortification strategies in dicot plants by intercropping with gramineous species. A review, Agron. Sustain. Dev. 29, 63–71.

### **Ants and Sustainable Agriculture**

Gero Benckiser

#### Contents

1	Intro	oduction	16
2	Fora	ging, Nutritional Behaviour, Intra-/	
	Inter	species Communication and Symbiont	
	Man	agement in Ant Colonies	16
	2.1	Nest Organisation and Functioning	17
	2.2	Foraging and Nutritional Aspects	17
	2.3	Bacterial Partners	18
	2.4	Fungal Farming	19
3	The	Low N-Input Concept of Harvester and Leaf	
	Cutt	er Ants	19
	3.1	Ants in Land-Use Mosaics	19
	3.2	Organic Matter Imports and Their Conversion .	20
	3.3	Human and Ant Farming – A Comparison	21
	3.4	Reduction of Carbon Decline and N <sub>2</sub> O	
		Emissions	22
4	Conc	clusion	22
Re	eferen	ces	23

G. Benckiser (🖂)

Abstract 60% of the world's ecosystems are not used in a sustainable way. Modern agriculture is blamed for declining soil carbon and biodiversity. Climate change, habitat fragmentation and other obstacles impede the movement of many animal species, and distribution changes are projected to continue. Therefore, we need alternative management strategies. The colony organisation of social insects, especially of ants, is seen as a model to design an improved agricultural management, because ants are very experienced agriculturists. Ants represent half of the global insect biomass. Their individuals work like a super organism. This article focuses on harvester and leaf cutter ants by considering Lasius species. It reviews the organisation structure of social ant communities. Harvester and leaf cutter ants represent a high percentage of the worldwide ant societies. They collect plant saps with carbon nitrogen (C/N) ratios of about 40 for their own nourishment and leaf fragments with C/N ratios of about 100 for fungi gardens and brood nourishment. They sustain huge numbers of individuals with their low N-based organic imports and their colony commensalisms enable them to convert these polymers into lower molecular, partly volatile compounds, adenosinetriphosphate (ATP), and heat. Digging improves water infiltration, drainage and soil aeration. Ants maintain fungi as a food source for the scleroproteinous brood, carry out food preservation, infection control and waste management, and construct with endurance new nests and rebuild them after damage. All these activities move the nest sites far away from the thermodynamic equilibrium. Physical, chemical and biological gradients emerge and the growing populations, together with nest-penetrating mycorrhized plant roots, absorb the released nutrients and form biomass

Department of Applied Microbiology, Justus-Liebig University, Giessen, Heinrich Buff-Ring 26-32, Germany e-mail: gero.benckiser@umwelt.uni-giessen.de

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_2,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Benckiser, Agron. Sustain. Dev. 30 (2010) 191–199. DOI: 10.1051/agro:2009026

by lowering energy flows into potentially strong consumer-resource interactions or runaway consumptions. The plant material import of leaf cutter ants, rich in carbon but low in proteins, amounts to 85-470 kg dry weight per year. It keeps the electron donor/acceptor ratio in favour of the electron donor so that denitrifiers can reduce nitrate predominantly to N<sub>2</sub>. Ants living in highly N-polluted areas bind the pollutant in the cuticle. In their low N-input environments harvester, leaf cutter and honeydew-sucking ants furnish the N demand of adult ants with the help of N<sub>2</sub>-fixing bacteria. The low N-input management of harvester, leaf cutter and honeydew-sucking ants is therefore a resourceful concept for approaching a highly productive agriculture by avoiding soil carbon decline and N<sub>2</sub>O emissions increase.

**Keywords** Ants • Colony organisation • Ecosystem engineers•Biotechnologists•Bio-indicators•Intra- and interspecies communication•Job sharing•Farming and waste management•Plasticity•Distribution in land-use mosaics•C/N management•N<sub>2</sub>O emissions

#### 1 Introduction

Nature's functioning depends upon microbial interactions among plants and animals, and their ability to adapt to environmental changes. Their relationship with each other and the environment relies on communication (signal exchanges), and the ability to assume dormancy (cryptobiosis) under unfavourable conditions (Benckiser and Schnell, 2007). The essential role ecosystems have is to provide services to humankind such as nutrient cycling, pest control, pollination, quality of life, and hydrological, atmospheric and climatic regulation. Among the macronutrients, nitrogen, a most unusual element, attracts a lot of interest because it interacts in various ways with ecosystem functioning.

Along its biogeochemical pathway nitrogen is reduced to  $NH_4^+$ , and oxidised into NO,  $NO_2^-$ ,  $N_2O$  and  $NO_3^-$ . It reaches by leaching, wind, erosion, or transport by animals any part of the Earth's system, no matter where it is introduced. The N cascade is interrupted only when nitrogen is incorporated into biomass, humus and clay minerals. The manifold nitrogen links to the environment make an effective control on farm and landscape level difficult. Instead, N<sub>2</sub> nitrogen increasingly returns as NH<sub>3</sub>, NOx and N<sub>2</sub>O into the atmosphere, where visibility decreases, the stratospheric ozone is depleted, and global warming and precipitation of acidity increase. Acidity in soils changes the biodiversity (Raubuch and Beese, 2005) and an alternative soil C/N management is required (Crutzen et al., 2007; Fargione et al., 2008; Searchinger et al., 2008).

This review portrays the self-organised living together predominantly of harvester, leaf cutter and honeydew-sucking ants, with their nest functioning based on organic imports high in carbon but low in nitrogen; and it is asked whether a reduction of soil carbon decline and greenhouse gas emissions could be approached by adapting their C/N management.

#### 2 Foraging, Nutritional Behaviour, Intra-/Interspecies Communication and Symbiont Management in Ant Colonies

Ants, order Hymenoptera, family Formicidae, play among insects a demonstrated role in transporting and pooling nutrients (Mueller et al., 2005). They are highly mobile, skilful tillers of soil, dispensers of seeds and microbial propagules, transmitters of N<sub>2</sub>-fixing bacteria, predators, ecosystem engineers, fungi growers, waste managers, biotechnologists, pest controllers, soldiers, and reproducers (Balasubramani and Kannaiyan, 1991; Jouquet et al., 2006; Philpott et al., 2006; Benckiser, 2007; Hölldobler and Wilson, 2009). Ants such as Atta colombica, A. laevigata, A. volenweideri and A. sexdens colonies comprise 1–2, 3, 5, 4–7, 5–8  $\times$  10<sup>6</sup> individuals, respectively, and are organised in clear patterns of colonisation and succession (probalistic patchiness). The myrmicine tribe Attini has developed during the last 50 million years a nest management based on organic imports high in carbon and low in protein, and gained expertise in fungal farming. Despite the manifold capabilities ants have, ecosystem models mostly neglect them (Moore et al., 2007; Brussard et al., 2007; Osler and Sommerkorn, 2007).
#### 2.1 Nest Organisation and Functioning

For constructing and repairing nests, which are built in a self-organised way without a central control - the queen is merely an egg-layer not an authority figure (Gordon, 2007) - ants must have at minimum the skill to find a suitable site, to evaluate several potential sites for choosing the best one, and to bring the entire population there safely (Hölldobler and Wilson, 1990; Pratt et al., 2002). In an Australian wheat/pasture rotation 5100 to 14860 ant nests per ha are constructed annually and 100 to 370 kg soil turned around, altering the soil structure, oxygen diffusion and metabolism of the nest environment as drastically as in conventionally treated agricultural sites where only a few Myrmica and Lasius species survive (Wang et al., 1995; Cannon, 1998; Lobry de Bryuyn, 1999; Dauber, 2001; Bucher and Marchesini, 2004; Wagner et al., 2004). Ant nests, reconstructed in late autumn so that they are smaller than at the height of summer, are damaged four times more by conventional agriculture than by no-tillage systems. Ants assemble for digging only when a certain signal threshold concentration is surpassed. The higher the group size, the more body contacts or chemical signals stimulate individuals to dig, and at an assembling volume plateau (V<sub>s</sub>) of around 0.2 spontaneous motor activity or lowdimensional disorder of individual ants translates into a directed mobility, physiological rhythm, or coordinated digging (Miramontes et al., 2001; Hölldobler and Wilson, 2009). With nest enlargement the signal concentration decreases and the digging frequency,  $\alpha$ , slows down. Ants not engaged in building activities cause feedbacks with snowballing effects on relocation and deposition of building blocks and only the colony level with its specific behavioural rules, activity controls, and high genetic and phenotypic plasticity (fitness), not the individual ant, has the global knowledge of the final nest structure, which tentatively ends in a circular form (Roces, 2002; Detrain and Deneubourg, 2006; Benckiser, 2007).

In conclusion, digging of ants changes soil structures as drastically as human activities (Hölldobler and Wilson, 2009). The mound distribution in a German meadow of *Lasius flavus*, a quince yellow, distributed throughout Europe, small-scale vegetation mosaics creating ant species (Fig. 1), shows that ant activities are not only significantly influencing the soil structure and vegetation in the nest surroundings, but as Fig. 1



Fig. 1 Distribution of *Lasius flavus* mounds in a meadow near the village Münzenberg, Hesse, Germany (photo Gero Benckiser)

suggests, in the whole meadow. The vegetation on the mounds is restricted to a few, mostly therophytic (annual) plant species and only a sub-sample of the surrounding grassland vegetation. More excessive is the digging of leaf cutter ants in South America, as the nest pictures in Hölldobler and Wilson (2009) reveal. The nest sizes of *Atta spp*. reach 26.1 to 67.2 square metres and soil depths down to 7 to 8 metres by excavating approximately 40 tons of soil per nest and constructing 6-m-long tunnels that connect an estimated harvesting area of more than one hectare.

#### 2.2 Foraging and Nutritional Aspects

Foraging, similarly coordinated to digging, starts early in the morning with a small group of patrollers, which leaves the nest, meanders around, and after successfully returning, stimulates other ants to join them for foraging (Alonso and Agosti, 2000; Detrain and Deneubourg, 2006). Polyphagous harvester and leaf cutter ants collect plant saps and/or solid leaf fragments with wide C/N ratios. Stressed plant parts are preferred, because they seemingly meet the food requirement of fungi gardens better (Currie et al., 1999; Meyer et al., 2006; Hölldobler and Wilson, 2009). Fungal staphylae, low in N, are the best balanced blends of nutritional components for the brood. Adult ants receive more than 90% of their energy requirement through imbibed plant saps, which are temporarily stored in the gut of returning ants and freely shared with intranidal individuals by regurgitation ('communal stomach'). During a day ants always return to the

same foraging area but the colony level alone can rapidly and at any time tune the right numbers of foragers to places of current food availability.

Acromyrmex spp. and Atta spp. subsist entirely on food low in nitrogen and abstain from supplementation by arthropod prey (D'Ettorre et al., 2001; Bucher and Marchesini, 2004; Hölldobler and Wilson, 2009). From the harvested solid food particles and plant saps, separated from each other by ant mouthparts, infrabuccal pockets and the proventriculus (Cannon, 1998), Acromyrmex subterraneus and A. crassispinus can hydrolyse cellulose, starch, maltose and sucrose while their fungal associates affect laminarin, xylan and the phenol fraction but not cellulose. Pogonomyrmex molefaciens and Messor pergandei collect seeds of grasses. At a low supply of desirable seeds they also consume less desirable seeds and non-seed plant material such as leaves. Honeydew-sucking Lasius niger and L. flavus species receive major proportions of nitrogen through intestinal, N<sub>2</sub>-fixing bacteria (Fischer et al., 2001; Zients et al., 2005) and Camponotus compressus stimulates seed germination to receive sugars and urea (Shetty, 1982). C. herculeanus circumvents N restrictions by spraying formic acid into wounds of preys and digesting proteinous macromolecules extraorally (Zhou et al., 2002). Foregut protease-lacking ants such as the fire ant, Solenopsis richteri Forel, avoid proteinrich diets (Ricks and Vinson, 1972) and queens contribute to nutrition by dispersing fungal symbionts from parent to offspring nests (Cannon, 1998; Green et al., 2002). Plant saps in the crop of Camponotus *pennsylvanicus* may contain 1208  $\pm$  169 µg sugars,  $5 \pm 0.7 \ \mu g$  glycogen and  $520 \pm 115 \ \mu g$  nitrogenous material (Cannon, 1998). Glycogen, presumably digested as it is consumed, is low-concentrated in the gut of C. pennsylvanicus similarly to lipids and lipoidal compounds (hormones, pheromones, antibiotic substances), which are crucial in colony recognition. In respect to lipids and lipoidal compounds it is assumed that they are directly shunted into postpharyngeal glands, having oily contents (Cannon, 1998; Zients et al., 2005), but some ants such as S. richteri Forel produce lipases in the mandibular and salivary glands, foregut, midgut, and hindgut that correlate with the "grease-loving" feeding habits of this insect (Ricks and Vinson, 1972).

In conclusion, ants can quickly recover from threats and nitrogen limitations by employing fungi and N<sub>2</sub>fixing symbionts, being carnivorous, and/or linking nutrition and parts of the regulatory network to the environment. Redundancy occurs only at a "critical minimum" of essential resources (commodity of space, habitat, right type of food or nest destruction through agriculture; Liebig's law, the principal tenet of population ecology; Fillman and Sterling, 1985; Showler et al., 1990; Zients et al., 2005; Hölldobler and Wilson, 2009).

#### 2.3 Bacterial Partners

Ants hatch endosymbiotic  $\alpha$ -,  $\beta$ -,  $\gamma$ -*Proteobacteria*, Flavobacteria and Actinomyces, which provide them with nitrogen, essential amino acids, vitamins, antibiotics and other supplements (Zients et al., 2005, 2006; Stoll et al., 2007; Hölldobler and Wilson, 2009). Blochmannia, a y3-subgroup Proteobacterium, resides in densely filled pouch-like ant structures (bacteriocytes) between the epithelial midgut cells and intestines, where it fixes N<sub>2</sub>. The Blochmannia genome, reduced to 450-800kb, controls N<sub>2</sub> fixation, the biosynthesis of essential amino acids, acetyl-CoA and tyrosine (required for cuticle sclerotisation), the urease that cooperates with the glutamine synthetase, and sulphate reduction. Despite lacking DNA repair genes the symbiosis between B. floridanus and Camponotus spp. is relatively stable and decays only when NH<sub>4</sub><sup>+</sup> accumulates in adult ants with finished sclerotisation. Another widespread  $\alpha$ -Proteobacterium among insects is Wolbachia, that dwells in the ovaries of worker ants and queens and has an influence on the rearing ratio of male and female ants. Wolbachia, closely affiliated with Rickettsae and Rhizobiales, cooperates with strains of the genera Escherichia, Salmonella, Yersinia, Bartonella, Pantoea, Sodalis Mesorhizobium, Agrobacterium and Blochmannia for the ants' benefit. Work and matter fluxes in the environment of ants and their microbial associates are coordinated by intra- and inter-exchanged mechanical and chemical signals, e.g., the height of honeydew, phloem saps, pheromones, increasing CO2 nest concentrations, various nonidentified substances released from a Dufour or poison gland, cuticular hydrocarbons, the nest temperature, which may vary in temperate regions with colonies of less than fifty workers between 16 and 21°C (Myrmica punctiventris Roger; M. rubra spp.) and in warmer regions with large colonies of  $10^6$  and more individuals between 23 and 32°C (*Formica polyactena*, *Camponotus mus*, *Solenopsis invicta*), 'thigmotaxis' and odometry (Way, 1963; Banschbach et al., 1997; Currie et al., 1999; Ruano et al., 2000; Benckiser, 2007; Bollazzi and Roces, 2007; Bollazzi et al., 2008; Hölldobler and Wilson, 2009). By modifying attenuation and degradation of signalling chemicals the nest temperature acts indirectly (Shaw et al., 2006).

## 2.4 Fungal Farming

High-input systems should focus more on yield with less fertiliser N, but even low-input systems require N to increase yield level and yield stability. Sustainable production systems on low N inputs have to be developed and to reach this goal cross-linked research on different scales, from single cells to crops, to diverse cropping and farming systems are required. In developing more environmentally-friendly agricultural systems ideas could be deduced from fungal farming, which is described here.

Lasius fulginosus cooperating with the fungus Cladosporium myrmecophilum and the ant farmers of the Attini genera Leucoagaricus and Leucocoprinus, which hatch basidiomycetes of the family Lepiotaceae, exemplify ant-fungi symbioses (Hölldobler and Wilson, 2009).

Concerning the quality of harvested plant material the ant and fungus have conflicting requirements and the decision is generally made in favour of the fungus, more or less irrespective of the attractiveness of the plant sap found during harvesting. From time to time ascomycetes of the family *Escovopis* infect basidiomycetal fungi farms (Mueller et al., 2005). Leaf cutter ants are successful in reacting to infection events by:

- (a) sequestration of the fungus garden from the environment,
- (b) an intensive monitoring of the fungus garden,
- (c) an early abatement of pathogens with filamentous antibiotic-producing bacteria of the genus *Pseudonorcardia* carried on the cuticle,
- (d) hatching an array of auxiliary microbes for disease suppression and other pro- and prebiotic services (Lopes and Orduz, 2003),
- (e) dispersing clonal fungi across many farmer generations,

- (f) maintaining reservoirs of genetically variable cultivars,
- (g) switching from time to time to novel cultivars,
- (h) sharing domesticated fungal cultivars with distantly related ant species,
- (i) a very specific and genetically predetermined feeding behaviour on mycelial fragments,
- (j) production of antibiotic substances in metapleural glands with different roles in fungus culturing (phenylacetic acid suppresses bacterial growth; myrmicacin, a hydroxydecanoic acid, inhibits the germination of spores of alien fungi; indol acetic acid, a plant hormone, stimulates mycelial growth, and 20 recently found compounds spanning in addition to keto acids, alcohols and lactones the whole range of carboxylic acids, from acetic acid to long-chain fatty acids), and finally,
- (k) a well-organised waste management accounting for approximately 10% of the work that accurately separates waste managers, consisting of a transporter caste, a worker caste and further partitions within the nest, from in- and outside moving ants with other duties, as well as
- (1) sensitive reactions of sympatric fungi growers on faecal droplets not derived from their own domesticated fungus. The ingredients of faecal droplets, used to manure newly established fungi gardens, help to identify undesirable basidiomycetal strains. Substrate and microbial imports, not harmful for the ant but for the fungus and less quickly identified, may be rejected too late and thus harm nest functioning (Herz et al., 2008).

# 3 The Low N-Input Concept of Harvester and Leaf Cutter Ants

#### 3.1 Ants in Land-Use Mosaics

Ants are competitors with each other. They live on honeydew, are predators, even cannibals, or collect within radii of less than one metre to a few hundred metres plant saps and leaf fragments with wide C/N ratios (Hölldobler and Wilson, 1990). Their densities and compositions in agricultural sites depend on human activities and are predictable in a typical German agricultural land-use mosaic such as the "Lahn-Dill Bergland"

with arable, fallow, grassland fields and forest sites (Table 1; Dauber, 2001; Braschler, 2005). Twentyseven ant species were found there between 1997 and 1998. They belonged to migrating forest species (Myrmica ruginodis, M. lobicornis, L. acervorum and Formica fusca), cursorial active species (F. rufibarbis, F. cunicularia, F. pratensis, F. polycatena, F. rufa, F. fusca, F. rufibarbis, F. cunicularia, F. pratensis, F. polyctena, F. rufa, F. sanguinea, M. ruginodis, M. lobicornis, Lasius platythorax and L. acervorum), peat land and humid meadow species (M. vandeli), thermophylic species (M. sabuleti, M. schencki, M. rugulosa, Tetramorium caespitum, Tapinoma erraticum and L. alienus), and relatively resistant species against agricultural vagaries (M. scabrinoides, M. rubra, M. ruginodis, L. flavus and L. niger), whereas in tropical countries harvester and leaf cutter ants, which can be agricultural pests, dominate the scene

**Table 1** Total and mean species richness of replicate sites and standard deviations of ants and four other taxa in different land

(Fillman and Sterling, 1985; Showler et al., 1990).

Hohenahr-Erda, Central Hesse, Germany. Different letters indicate that values are significantly different (P < 0.05, unequal N HSD-Test; for details see (Dauber et al., 2005)

use types sampled with pitfall traps in the landscape mosaic of HSD-Test; for details see (Dauber et al., 2005)				
		Arable land	Grassland	Fallow land
		(n = 12)	(n = 12)	(n = 10)
Carabids	total	75	57	65
	mean	$26.6 \pm 3.7a$	$19.3 \pm 4.9b$	$20.4 \pm 4.6b$
Staphylinids	total	71	70	97
	mean	$18.8 \pm 6.1a$	$21.7 \pm 7.0a$	$33.3 \pm 6.3b$
Ants	total	17	17	21
	mean	$6.3 \pm 2.5a$	$7.0 \pm 3.4a$	$9.1 \pm 3.2b$
Isopods	total	6	10	12
	mean	$1.1 \pm 1.0a$	$2.1 \pm 0.8 b$	$3.0 \pm 1.5c$
Diplopods	total	9	10	11
	mean	$2.0 \pm 1.2a$	$2.7 \pm 1.4a$	$4.0 \pm 0.9 \mathrm{b}$



# 3.2 Organic Matter Imports and Their Conversion

Harvester ants collect plant saps and transport them in the crop to the nest to feed non-foraging nest mates by regurgitation. In one g of collected plant saps 542 to 718 mg carbon, 13.8 to 20.5 mg nitrogen and 58.0 to 94.0 mg ash contents have been determined (Cannon, 1998). The C/N ratios of plant saps and leaf particles, also transported into the nests, are between 40 and 100. The anabolic-catabolic nest cascade (Fig. 2) narrows them to 37-25 (refuse material). Thereby cellulose, fibre, phenol, true protein and the insoluble N fractions may decrease by about 47.7, 9.1, 26.1, 60.8, 10.9 and 10.9%, respectively, while lignin, crude protein, ash content, total N, soluble N, and soluble inorganic and organic N may increase by about 4.9, 31.9, 324.6, 31.9, 363.2, 150 and 461.5%, respectively. Digging in harvester ant nests enhanced the ash content of the

deposited refuse material by 71-74.6%, in which total nitrogen increased by 5.2-17% and total carbon decreased by 10.1-34.3% (Cannon, 1998; Wagner and Jones, 2006). The soil chemistry in leaf cutter nests of the 24 known Acromyrmex and 15 known Atta spp. may alter similarly to in the harvester nests. Volatile organic intermediates (Vespermann et al., 2007),  $CO_2$  and numerous other signal compounds, released into the nest environment, coordinate there the soil functioning (digging, foraging, waste management and infection avoidance). Nest-penetrating, mycorrhized plant roots and the growing nest commensalisms consisting of multiplying microbes, the cuticle sclerotising brood, and adult ants absorb the available nutrients, lower the energy flows into potentially strong consumer-resource interactions or runaway consumptions and develop super organisms (Ben-Jacob, 2003; Hölldobler and Wilson, 2009). The continuous organic imports into harvester and leaf cutter nests, characterised by C/N ratios between 40 and 100, keep the electron donor/acceptor, carbohydrate/nitrate ratio in favour of the electron donor (Jones and Wagner, 2006). At such carbohydrate/nitrate ratios denitrifiers of the nest environment will reduce accumulating nitrate to N<sub>2</sub>, as field studies in plots, which have received varying amounts of mineral and organic fertiliser, and studies in other environments have shown (Tiedje, 1988; Simarmata et al., 1993; Ratering et al., 2007; Falkowski et al., 2008). N<sub>2</sub>O emissions are lowest at a carbon/nitrate-N ratio of about 50 and from earthworms, which prefer to ingest organic matter with narrow C/N ratios, it is known that comparably high N<sub>2</sub>O emissions can be released (Rizhiya et al., 2007).

# 3.3 Human and Ant Farming – A Comparison

Farmers, being competitive on the global market, apply high rates of mineral N fertilisers to strongly reduced, fully mechanised and pesticide-treated crop rotations, increasingly grown on large sized fields with various soil types, slopes and valleys. Fertilisation brought a quantum leap in yield improvement, yet concomitantly

narrows the C/N ratio and, e.g., increasing amounts of N<sub>2</sub>O are emitted from palm- and soybean-biodiesel farms with soil carbon debts of 702 to 3452 Mg CO<sub>2</sub>  $ha^{-1}$  (Crutzen et al., 2007; Fargione et al., 2008; Searchinger et al., 2008). A better carbon/nitrogen management is required (Schmidt-Rohr et al., 2004; Honermeier, 2007). Approximately 50 to 60 million years ago and long before the Romans, who perceived the importance of the C/N ratio and started to culture certain crops and trees with critical microbial rhizosphere associates (nitrogen-fixing bacteria, mycorrhizal fungi), ants began to domesticate fungi and to control the carbohydrate/nitrate ratio in their nest environment (Hölldobler and Wilson, 1990). Today's research ties in with this long tradition and investigates more detailed root exudateconsuming, antibiotic-producing, disease-suppressant, plant health-amending bacteria and fungi (Benckiser, 1997; Gómez-Gómez and Boller, 2002; Lopes and Orduz, 2003; Franken and George, 2007), soil carbon stabilisation (Von Lützow et al., 2007), N-demandnet-N-mineralisation-gross-N-release balancing (Icoz and Stotzky, 2008; Kooijman et al., 2008), terra preta (Marris, 2006), extended crop rotations (e.g., red clover-red clover-potato-wheat-field beans-wheatrye or corn field beans-wheat-barley-clover/grassclover/grass; Honermeier, 2007; Malezieux et al., 2009), the harvester and leaf cutter-CN-nest management (Jones and Wagner, 2006; Wagner and Jones, 2004), soil worm behaviour (Sticht et al., 2006; Rizhiya et al., 2007), recuperation of shifting cultivation systems (Addiscott, 1995), and the domestication of low N-input varieties (Ruiz et al., 2008). Recent findings show that in extended crop rotations with legume fields the nitrogenase, which stimulates soil phosphatases, affords a substantially lower activation energy ( $\sim 103 \text{ kJ mol}^{-1}$ ) than the nitrogenase enzyme itself ( $\sim 210 \text{ kJ mol}^{-1}$ ) and that N<sub>2</sub>-fixing plants and ants, which dispense N<sub>2</sub>-fixing bacteria among legumes (Balasubramani and Kannaiyan, 1991), have a similar geographical (temperature-dependent) distribution (Banschbach et al., 1997; Houlton et al., 2008). Whether the activation energy of the nitrogenase in N<sub>2</sub>fixing bacteria of honeydew-sucking ants or in organically fertilised fields also needs substantially lower activation energy than the nitrogenase enzyme itself has to be proven.

# 3.4 Reduction of Carbon Decline and N<sub>2</sub>O Emissions

Nowadays farmers try to reduce soil carbon decline and  $N_2O$  emissions by converting conventionally treated fields into organic farming systems or by employing fertilisers with nitrification inhibitors such as dicyandiamide (DCD), nitrapyrin or 3.4dimethylpyrazole phosphate (DMPP) (Duxbury et al., 1982; Nelson and Huber, 2001; Honermeier, 2007). Nitrification inhibitors should only temporarily block the ammonium monooxygenase and denitrification and not impair other essential soil processes. A study in DMPP- and DCD-N-fertilised, unfertilised, and control field plots over 3 years could not fully exclude that nitrification inhibitors impair essential soil processes (Weiske et al., 2001); and accordingly fresh ideas from the low N-input management of harvester and leaf cutter ant nests, which live in 26–67 m<sup>2</sup> sized nests with subterranean tunnels, ducts, fungus chambers and residue dumps, and where millions of individuals are digging, doing indoor farming, pest control and waste management, are of interest, though farmers know from agricultural tradition and experience that habitat-specific microbial communities beyond carbon sequestration and growth promotion can rapidly convert into detrimental ones and destabilise yields in various ways (Young and Crawford, 2005). Besides the difficult management of biological systems a substantial proportion of variation in species richness and functioning is still statistically explained in terms of a few environmental variables (Brussard et al., 2007; Moore et al., 2007; Osler and Sommerkorn, 2007) and a predictive theory for messaging molecules, signal transduction, integrated genome regulation and synergisms in subsystems is at the very beginning (Jones et al., 2006; Detrain and Deneubourg, 2006). However, these areas are a challenge for biologists, agriculturists, physicists, (bio-) chemists and ecologists and may prove to generate exciting results about the manipulative power that synergistic systems can exert over each partner (Hölldobler and Wilson, 2009).

## 4 Conclusion

Harvester and leaf cutter ants keep invaders low by antibiotics and by importing organic material, low in



**Fig. 3** Ant activities and possible exertions of influence on dissolved organic matter (DOM), inorganic and organic N pools, CO<sub>2</sub> production, and gross N fluxes

nitrogen but rich in carbon, by storing nitrogen temporarily in the cuticle and by keeping the nest C/N ratio in favour of the electron donator (Fig. 3).

There are few existing data, and more detailed studies are imperative, but let us speculate that the N2O/N2 ratio under leaf cutter nest conditions is wide and N<sub>2</sub>O is predominantly reduced to N<sub>2</sub>. Harvester and leaf cutter nest sites produce on a low N-input basis high biomasses, similarly to minerally fertilised agricultural soils, and accordingly such environments are seen as an ideal study ground for receiving hints how highly productive, conventional agricultural systems such as biofuel farms could avoid carbon decline and minimise greenhouse gas emissions (Benckiser, 2007; Hölldobler and Wilson, 2009). Organic farming systems, on the other hand, could learn from ant environments how their soil C and N management could be improved and food webs better used for pest control as Fig. 4 reveals.

Figure 4 exemplifies by the cooperation of stinging tree ants, genus *Crematogaster*, which hatch in their nests made from leaves and fixed between

Fig. 4 Synergisms between Rufous Woodpeckers, *Micropter*nus brachyurus (a), ants of the genus *Crematogaster* (b) and plant juices sucking mealybugs, family *Pseudococcidae* (c),

twigs mealybugs, family Pseudococcidae, with the medium sized, brown Rufous Woodpecker, Micropternus brachyurus, how agro-ecological management could progress in its development (Vishnudas, 2008). Mealybugs living in moist, warm climates, protected by a secreted powdery wax layer, suck plant juices from a variety of subtropical trees (Gymnocladus dioicus, Dalbergia latifolia, Aporusa lindleyana, Erythrina indica, Grevillea parallela, Mallotus alba, C. verum, Cinnamomam malabatrum, Olea dioica, Gliricidia indica, Hopea parviflora, Terminalia bellarica, Bischofia javanica, Syzygium cumini and Lagestroemia microcarpa). The ants use the honeydewproducing potential of mealybugs nutritionally and foster and protect them in their paper-wasp-like nests. The plant juice sucking mealybug being supported by Crematogaster ssp. impairs coffee plantations and both considered as pests and killed with pesticides.

The emerging demand for organic shade-grown coffee all over the world is forcing planters to reduce the use of toxic chemicals in coffee plantations. This helps maintain reasonable numbers of *Crematogaster* ant colonies and, as followers, many forest-dwelling

which all reside in the tree-nest (**d**), made of leaves. (the ant nest-photo, Top Slip, Western Ghats, Kerala, South India, Gero Benckiser; bird-, ant- and mealybug-photos, Wikepedia)

species such as the Rufous Woodpecker. The high conservation value of shaded coffee plantations with their diversity of canopy tree species providing critical habitats would be maintained and this is recognised. Proper agro-ecological management practices are starting to develop and are being popularised amongst planters.

#### References

- Addiscott T.M. (1995) Entropy and sustainability, Eur. J. Soil Sci. 46, 161–168.
- Alonso L.E., Agosti D. (2000) Biodiversity studies, monitoring, and ants: an overview, in: Agosti D., Maijer J.D., Alonso L.E., Schultz T.R., Ants – Standard methods for measuring and monitoring biodiversity, Smithonsian Institution Press, Washington, pp. 1–8.
- Balasubramani G., Kannaiyan S. (1991) Natural spread of A. caulinodans in stem nodulating S. rostrata, INSURF Netwatch 1, 5–6.
- Banschbach V.S., Levit N., Herbers J.M. (1997) Nest temperatures and thermal preferences of a forest ant species: is seasonal polydomy a thermoregulatory mechanism? Insect. Soc. 44, 109–122.

G. Benckiser

- Ben-Jacob E. (2003) Bacterial self-organization: coenhancement of complexification and adaptability in a dynamic environment, Philos. T. Math., Phys. Eng. Sci. 361, 1283–1312.
- Benckiser G. (1997) Organic inputs and soil metabolism, in: Benckiser G. (Ed.), Fauna in soil ecosystems – recycling processes, nutrient fluxes, and agricultural production. Marcel Dekker, New York, pp. 7–62.
- Benckiser G. (2007) Principles behind order and sustainability in natural successions and agriculture, in: Benckiser G., Schnell S. (Eds.), Biodiversity in agricultural production systems. CRC, Taylor and Francis, Boca Raton, Florida, pp. 349–383.
- Benckiser G., Schnell S. (Eds.) (2007) Biodiversity in agricultural production systems. CRC, Taylor and Francis, Boca Raton, USA.
- Bollazzi M., Roces F. (2007) To build or not to build: circulating dry air organizes collective building for climate control in the leaf-cutting ant *Acromyrmex ambiguous*, Anim. Behav. 74, 1349–1355.
- Bollazzi M., Kronenbitter J., Roces F. (2008) Soil temperature, digging behaviour, and the adaptive value of nest depth in South American species of *Acromyrmex* leaf-cutting ants, Oecologia 158, 165–175.
- Braschler B.M. (2005) Effects of experimental small-scale grassland fragmentation on the population dynamics of invertebrates, Ph.D. thesis, University of Basel, Switzerland.
- Brussard L., de Ruiter P.C., Brown G.G. (2007) Soil biodiversity for agricultural sustainability, Agr. Ecosyt. Environ. 121, 233–244.
- Bucher E.H., Marchesini V. (2004) Herbivory by leaf-cutting ants: nutrient balance between harvested and refuse material, Biotropica. 36, 327–332.
- Cannon C.A. (1998) Nutritional Ecology of the Carpenter Ant Camponotus pennsylvanicus (De Geer): Macronutrient Preference and Particle Consumption. Dissertation submitted to the Faculty of the Virginia Polytechnic Institute and State University in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Entomology, Blacksburg, Virginia scholar lib.vt.edu/theses/ available/ etd-110398-001812/ unrestricted/Dissertation-etd.PDF.
- Currie C.R., Scott J.A., Summerbell R.C., Malloch D. (1999) Fungus-growing ants use antibiotic producing bacteria to control garden parasites, Nature 398, 701–704.
- Crutzen P.J., Mosier A.R., Smith K.A., Winiwarter W. (2007)  $N_2O$  release from agro-biofuel production negates global warming reduction by replacing fossil fuels, Atm. Chem. Phys. Discus. 7, 11191–112055.
- Dauber J. (2001) Ant communities of an agricultural landscape: Relationships to landscape structure and land-use management, Ph.D. thesis, Justus Liebig- University of Giessen, Germany, 120 p.
- Dauber J., Purtauf T., Allspach A., Frisch J., Voigtländer K., Wolters V. (2005) Local vs. landscape controls on diversity: a test using surface-dwelling soil macro-invertebrates of differing mobility, Global Ecol. Biogeogr. 14, 213–221.
- D'Ettorre P., Mora P., Dibangou V., Rouland C., Errard, C. (2001) The role of the symbiotic fungus in the digestive metabolism of two species of fungus-growing ants, J. Comp. Physiol. B 172, 169–176.

- Detrain C., Deneubourg J.L. (2006) Self-organized structures in a superorganism: do ants "behave" like molecules? Phys. Life Rev. 3, 162–187.
- Duxbury J.M., Bouldin D.R., Terry R.E., Tate III R.L. (1982) Emissions of nitrous oxide from soils, Nature 298, 462–464.
- Falkowski P.G., Fenchel T., Delong E.F. (2008) The microbial engines that drive earth's biochemical cycles, Science 320, 1034–1039.
- Fargione J., Hill J., Tilman D., Polasky S., Hawthrone P. (2008) Land clearing and the biofuel carbon debt. Science 319, 1235–1238.
- Fillman D.A., Sterling W.L. (1985) Inactivation Level for the red imported fire ant, *Solenopsis invicta (Hym: Formicidae)*: a predator of the boll weevil, anthomonas grande (Col.: Curculionidae), Agr. Ecosyst. Environ. 13, 93–122.
- Fischer M.K., Hoffmann K., Völkl W. (2001) Competition for mutualists in an ant-homopteran interaction mediated by hierachies of ant attendance, Oikos 92, 531–541.
- Franken P., George E. (2007) Diversity of arbuscular mycorrhizal fungi, in: Benckiser G., Schnell S. (Eds.), Biodiversity in agricultural production systems, CRC, Taylor and Francis, Boca Raton, Florida, pp. 189–203.
- Gómez-Gómez L., Boller T. (2002) Flagellin perception: a paradigm for innate immunity, Trends Plant Sci. 7, 251–256.
- Gordon D.M. (2007) Control without hierarchy, Nature 446, 1433.
- Green A.M., Mueller U.G., Adams R.M.M. (2002) Extensive exchange of fungal cultivars between sympatric species of fungus-growing ants, Mol. Ecol. 11, 191–195.
- Herz H., Hölldobler B., Roce F. (2008) Delayed rejection in a leaf-cutting ant after foraging on plants unsuitable for the symbiotic fungus, Behav. Ecol. 19, 575–582
- Hölldobler B., Wilson E.O. (1990) The ants. Springer, Heidelberg, 732 p.
- Hölldobler B., Wilson E.O. (2009) The superorganism, W.W. Norton & Company, Inc., New York, 522 p.
- Honermeier B. (2007) Diversity in crop production systems, in: Benckiser G., Schnell S. (Eds.), Biodiversity in Agricultural Production Systems, CRC, Taylor & Francis, Boca Raton, pp. 1–19.
- Houlton B.Z., Wang Y.P., Vitousek P.M., Field C.B. (2008) A unifying framework for dinitrogen fixation in the terrestrial biosphere, Nature 454, 327–331.
- Icoz I., Stotzky G. (2008) Fate and effects of insect-resistant *Bt* crops in soil ecosystems, Soil Biol. Biochem. 40, 559–586.
- Jones C.G., Gutiérrez J.L., Groffman P.M., Shachak M. (2006) Linking ecosystem engineers to soil processes: a frame work using the Jenny State Factor Equation, Eur. J. Soil Biol. 42, 39–53.
- Jones J.B., Wagner D. (2006) Microhabitat-specific controls on soil respiration and denitrification in the Mojave Desert: the role of harvester ant nests and vegetation, West. N. Am. Naturalist 66, 426–432.
- Jouquet P., Dauber J., Lagerlöf J., Lavelle P., Lepage M. (2006) Soil invertebrates as ecosystem engineers: Intended and accidental effects on soil and feed back loops, Appl. Soil Ecol. 32, 153–164.
- Kooijman A.M., Kooijman-Schouten M.M., Martinez-Hernandez G.B. (2008) Alternative strategies to sustain N-fertility in acid and calcaric beech forests: low microbial

N-demand versus high biological activity, Basic Appl. Ecol. 9, 410–421.

- Lobry de Bryuyn L.A. (1999) Ants as bioindicators of soil function in rural environments, Agr. Ecosyst. Environ. 74, 425–441.
- Lopes E., Orduz S. (2003) Metarhizium anisopliae and Trichoderma viride for control of nests of the fungus-growing ant, Atta cephalotes, Biol. Control. 27, 194–200.

Marris M. (2006) Black is the new green, Nature 442, 624-626.

- Malézieux E., Crozat Y., Dupraz C., Laurans M., Makowski D. Ozier-Lafontaine H., Rapidel B., de Tourdonnet S., Valantin-Morison M. (2009) Mixing plant species in cropping systems: concepts, tools and models. A review, Agron. Sustain. Dev. 29, 43–62.
- Meyer S.T., Roces F., Wirth R. (2006) Selecting the drought stressed: effects of plant stress on intraspecific and withinplant herbivory patterns of the leaf-cutting ant *Atta colombica*, Funct. Ecol. 20, 973–981.
- Miramontes O., Sole R.V., Goodwin B.C. (2001) Neural networks as sources of chaotic motor activity in ants and how complexity develops at the social scale, Int. J Bifurcat. Chaos 11, 1655–1664.
- Moore J.C., Simpson R.T., McCann K.S., de Ruiter P.C. (2007) Food web interactions and modeling, in: Benckiser G., Schnell S. (Eds.), Biodiversity in Agricultural Production Systems, CRC, Taylor & Francis, Boca Raton, pp. 385–398.
- Mueller U.G., Gerardo N.M, Aanen D.K., Six D.L., Schultz D.R. (2005) The evolution of agriculture in insects, Annu. Rev. Ecol. Sytst. 36, 563–595.
- Nelson D.W., Huber D. (2001) Nitrification inhibitors for corn production. National Corn Handbook, Iowa State University, USA, University extension, http://www.extension. iastate.edu/Publications/NCH55.pdf.
- Osler G.H.R., Sommerkorn M. (2007) Toward a complete soil C and N Cycle: Incorporating the soil fauna, Ecology 88, 1611–1621.
- Philpott S.M., Uno S., Maldonado J. (2006) The importance of ants and high-shade management to coffee pollination and fruit weight in Chiapas, Mexico. Biodiv. Conser. 15, 487–501.
- Pratt S.C., Mallon E.B., Sumpter D.J.T., Franks R.N. (2002) Quorum sensing, recruitment, and collective decisionmaking during colony emigration by the ant *Leptothorax albipennis*, Behav. Ecol. Sociobiol. 52, 117–127.
- Ratering S., Benckiser G., Schnell S. (2007) Metabolic diversity of micro-organisms in agricultural soils In: Benckiser G., Schnell S. (Eds.), Biodiversity in Agricultural Production Systems, CRC, Taylor & Francis, Boca Raton, pp. 287–315.
- Raubuch M., Beese F. (2005) Influence of soil acidity on depth gradients of microbial biomass in beech forest soils, Eur. J. Forest. Res. 124, 87–93.
- Ricks B.L., Bradleigh Vinson S. (1972) Digestive enzymes of the imported fire ant *Solenopsis richteri* (Hymenoptera: Formicidae), Entomol. Exp. Appl. 15, 329–334.
- Rizhiya E., Bertora C., van Vliet P.C.J., Kuikman P.J., Faber J.H., van Groenigen J.W. (2007) Earthworm activity as a determinant for N<sub>2</sub>O emission from crop residue, Soil Biol. Biochem. 39, 2058–2069.
- Roces F. (2002) Individual complexity and self-organisation in foraging by leaf-cutting ants, Biol. Bull. US 202, 306–313.

- Ruano F., Tinaut A., Soler J.J. (2000) High surface temperature select for individual foraging in ants, Behav. Ecol. 11, 396–404.
- Ruiz M., Aguiriano E., Carrillo J.M. (2008) Effects of N fertilization on yield for low-input production in Spanish wheat landraces (*Triticum turgidum* L. and *Triticum monococcum* L.), Plant Breeding 127, 20–23.
- Schmidt-Rohr K., Mao J.D., Olk D.C. (2004) Nitrogen-bonded aromatics in soil organic matter and their implications for a yield decline in intensive rice cropping, Proc. Natl. Acad. Sci. (USA) 101, 6351–6354.
- Searchinger T., Heimlich R., Houghton R.A., Dong F., Elobeid A., Fabiosa J., Tokgoz S., Hayes D., Yu T.H. (2008) Use of U.S. croplands for biofuels increases greenhouse gases through emissions from land-use change, Science 319, 238–240.
- Shaw L.J., Morris P. Hooker J. E. (2006) Perception and modification of plant flavonoid signals by rhizosphere microorganisms, Environ. Microbiol. 8, 1867–1880.
- Shetty P.S. (1982) Gustatory preferences of ants (*Camponotus compressus*) for urea and sugars, Experentia 38, 259–260.
- Showler A.T., Knaus R.M., Reagan T.E. (1990) Studies of the territorial dynamics of the red imported Fire Ant (*Solenopsis invicta* Buren, *Hymenoptera: Formicidae*), Agr. Ecosyst. Environ. 30, 97–105.
- Simarmata T., Benckiser G., Ottow J.C.G. (1993) Effect of an increasing carbon:nitrate ratio on the reliability of acetylene in blocking the N<sub>2</sub>O-reductase activity of denitrifying bacteria in soil, Biol. Fert. Soils 15, 107–112.
- Sticht C., Schrader S., Giesemann A., Weigel H.J. (2006) Effects of elevated atmospheric CO<sub>2</sub> and N fertilization on abundance, diversity and C-isotopic signature of collembolan communities in arable soil, Appl. Soil Ecol. 34, 219–229.
- Stoll S., Gadau J., Gross R., Feldhaar H. (2007) Bacterial microbiota associated with ants of the genus *Tetraponera*, Biol. J. Linn. Soc. 90, 399–412.
- Tiedje J.M. (1988) Ecology of denitrification and dissimilatory nitrate reduction to ammonium, in: Zehnder A.J.B. (Ed.), *Biology of anaerobic microorganisms*, John Wiley, New York, pp. 179–244.
- Vespermann A., Kai M., Piechullaet B. (2007) Rhizobacterial volatiles affect the growth of fungi and *Arabidopsis thaliana*, Appl. Environ. Microb. 73, 5639–5641.
- Von Lützow M., Kögel-Knabner I., Ekschmitt K. Flessa H., Guggenberger G., Matzner E., Marscner B. (2007) SOM fractionation methods: Relevance to functional pools and to stabilization mechanisms, Soil Biol. Biochem. 39, 2183–2207.
- Vishnudas C.K. (2008) Woodpeckers and ants in India's shade coffee, Indian Birds 4, 9–11.
- Wagner D., Jones J.B. (2004) The contribution of harvester ant nests (*Pogonomyrmex rugosus*, *Hymenoptera*, *Formicidae*) to soil nutrient stocks and microbial biomass in the Mojave Desert. Environ. Entomol. 33, 599–607.
- Wagner D., Jones J.B. (2006) The impact of harvester ant nests on decomposition, N mineralization, litter quality, and availability of N to plants in the Mojave Desert, Soil Biol. Biochem. 38, 2593–2601.

- Wagner D., Jones J.B., Gordonet D.M. (2004) Development of ant harvester colonies alters soil chemistry, Soil Biol. Biochem. 36, 797–804.
- Wang D., McSweeney K., Lowery B., Norman J.M. (1995) Nest structure of ant *Lasius neoniger* Emery and its implication to soil modification, Geoderma 66, 259–272.
- Way M.J. (1963) Mutualism between ants and honeydewproducing *Homoptera*, Ann. Rev. Entomol. 8, 307–344.
- Weiske A., Benckiser G., Herbert T., Ottow J.C.G. (2001) Influence of thenitrification inhibitor 3,4-dimethylpyrazole phosphate (DMPP) in comparison to dicyandiamide onnitrous oxide emissions, carbon dioxide fluxes and methane oxidation during 3 years of repeated application in field experiments, Biol. Fert. Soils 34, 109–117.
- Young I.M., Crawford J.W. (2004) Interactionsand selforganisation in the soil-microbe complex, Science 304, 1634–1637.
- Zhou X., Haya H.K., Heungens K., Goodrich-Blair H. (2002) Response of ants to a deterrent factor(s) produced by the symbiontic bacteria of Entomopathogenic nematodes, Appl Environ. Microb. 68, 6202–6209.
- Zients E., Feldhaar H., Stoll S., Gross R. (2005) Insights into the microbial world associated with ants, Arch. Microbiol. 184, 199–206.
- Zients E., Beyaert I., Gross R., Feldhaar H. (2006) Relevance of the endosymbiosis of *Blochmannia floridans* and carpenter ants at different stages of the life cycle of the host, Appl. Environ. Microb. 72, 6027–6033.

# Agroecology as a Science, a Movement and a Practice

A. Wezel, S. Bellon, T. Doré, C. Francis, D. Vallod, and C. David

## Contents

1	Intro	oduction	28
2	<b>Two</b> 2.1 2.2	Major Historical Periods of Agroecology The 'Old Age' of Agroecology: 1930s–1960s Expansion of Agroecology: 1970s–2000s	28 28 30
3	Exar	nples from Different Countries	31
	3.1	Case Study of the USA	31
	3.2	Case Study of Brazil	32
	3.3	Case Study of Germany	34
	3.4	Case Study of France	35
4	Disc	ussion	36
	4.1	Today's Scientific Discipline of Agroecology:	
		A Variation in Definitions and Scales	36
	4.2	Agroecology as a Movement or a Practice	38
	4.3	Combined Use of the Three Definitions	39
	4.4	Agroecology, a History of Oppositions?	- 39
	4.5	Future Questions and Challenges of the	
		Scientific Discipline of Agroecology	40
5	Cone	clusions	40
Re	eferen	ces	41

e-mail: wezel@isara.fr

Abstract Agroecology involves various approaches to solve actual challenges of agricultural production. Though agroecology initially dealt primarily with crop production and protection aspects, in recent decades new dimensions such as environmental, social, economic, ethical and development issues are becoming relevant. Today, the term 'agroecology' means either a scientific discipline, agricultural practice, or political or social movement. Here we study the different meanings of agroecology. For that we analyse the historical development of agroecology. We present examples from USA, Brazil, Germany, and France. We study and discuss the evolution of different meanings agroecology. The use of the term agroecology can be traced back to the 1930s. Until the 1960s agroecology referred only as a purely scientific discipline. Then, different branches of agroecology developed. Following environmental movements in the 1960s that went against industrial agriculture, agroecology evolved and fostered agroecological movements in the 1990s. Agroecology as an agricultural practice emerged in the 1980s, and was often intertwined with movements. Further, the scales and dimensions of agroecological investigations changed over the past 80 years from the plot and field scales to the farm and agroecosystem scales. Actually three approaches persist: (1) investigations at plot and field scales, (2) investigations at the agroecosystem and farm scales, and (3) investigations covering the whole food system. These different approaches of agroecological science can be explained by the history of nations. In France, agroecology was mainly understood as a farming practice and to certain extent as a movement, whereas the corresponding scientific discipline was agronomy. In Germany, agroecology has a long tradition as a

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_3,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Wezel et al., Agron. Sustain. Dev. 29 (2009) 503–515. DOI: 10.1051/agro:2009004

A. Wezel (🖂)

ISARA, Department of Agroecosystems, Environment and Production, 23 rue Jean Baldassini, 69364 Lyon Cedex 07, France

scientific discipline. In the USA and in Brazil all three interpretations of agroecology occur, albeit with a predominance of agroecology as a science in the USA and a stronger emphasis on movement and agricultural practice in Brazil. These varied meanings of the term agroecology cause confusion among scientists and the public, and we recommend that those who publish using this term be explicit in their interpretation.

**Keywords** Agroecosystem • Agronomy • Ecology • Food system • Rural development • Scientific discipline • Sustainable agriculture • Systems approach

#### 1 Introduction

The term 'agroecology' was first used in two scientific publications by Bensin (1928, 1930), and most recently in books by Gliessman (2007) and Warner (2007a). Between these dates, there are 80 years of history and confusion around definitions. The term agroecology has been used increasingly in scientific literature in recent years. For instance, the occurrence of the root agroecolog\* or agro-ecolog\* in the Web of Science increased from six in 1991 to 141 in 2007. The figures in a CAB abstracts search include two records in 1971; 102 in 1991; and 208 in 2007. This clearly indicates a great potential source of new information and perspective in agriculture and food systems. The term agroecology is currently used with quite different meanings in science, and also as describing a movement or agricultural practices. We recognise that confusion around the world concerning the term agroecology is partly due to translations, but also to meanings among and within different cultures. Our objectives are to explore and discuss three major uses of the term agroecology: science, movement and practice. Based on the historical development of agroecology in various time periods and geographical contexts, we discuss the evolution of its different meanings with examples from Europe (France and Germany), North America (USA), and South America (Brazil). To complete this analysis of the history of agroecology as a scientific discipline, we also consider some important reports that cover similar issues but where the term agroecology is not explicitly mentioned. The intent is to clarify use of the

word "agroecology" and to urge colleagues to be explicit in definition when the term is used in the future.

## 2 Two Major Historical Periods of Agroecology

## 2.1 The 'Old Age' of Agroecology: 1930s-1960s

The word "agroecology" emerged at the beginning of the 20th century. Thereafter, both its definition and scope evolved significantly. This path primarily can be related to the evolution of two disciplines from which agroecology is derived, agronomy and ecology, but also to other disciplines such as zoology and botany/plant physiology, and their applications in agricultural and environmental issues. The term agroecology or agroecological was firstly used by Bensin (1928, 1930, 1935) (Table 1), a Russian agronomist, who suggested the term 'agroecology' to describe the use of ecological methods in research on commercial crop plants (Bensin, 1930 cited in Klages, 1942). Agroecology would hence be preliminarily defined as the application of ecology in agriculture - a meaning which is still used.

In the 1950s, the German ecologist/zoologist Tischler (1950, 1953, 1959, 1961) published several articles in which he used the term agroecology. He presented results of agroecological research, in particular on pest management, and discussed unsolved problems concerning soil biology, insect biocoenosis interactions and plant protection in agricultural landscapes, including also non-cultivated habitats. His book was probably the first to be actually titled 'agroecology' (Tischler, 1965). He analysed the different components such as plants, animals, soils, and climate, and their interactions within an agroecosystem as well as the impact of human agricultural management on these components. This approach combines ecology, especially the interactions among biological components at the field or agroecosystem level, and agronomy with a focus on the integration of agricultural management.

Between the 1930s and 1960s related studies were published, without using the word agroecology in the

Table 1	Important	works in the	history	of Agroecolo	gy (adapted	from (	Gliessman, 20	<b>JO</b> 7)

Year	Author	Title	
1928	Bensin	Agroecological characteristics description and classification of the local corn varieties chorotypes <sup>1</sup>	
1928	Klages	Crop ecology and ecological crop geography in the agronomic curriculum <sup>2</sup>	
1930	Bensin	Possibilities for international cooperation in agroecological investigations <sup>2</sup>	
1930	Friederichs	Die Grundfragen und Gesetzmäßigkeiten der land- und forstwirtschaftlichen Zoologie <sup>1</sup>	
1938	Papadakis	Compendium on crop ecology	
1939	Hanson	Ecology in agriculture <sup>2</sup>	
1942	Klages	Ecological crop geography <sup>1</sup>	
1950	Tischler	Ergebnisse und Probleme der Agrarökologie <sup>2</sup>	
1956	Azzi	Agricultural ecology <sup>1</sup>	
1965	Tischler	Agrarökologie <sup>1</sup>	
1967	Hénin	Les acquisitions techniques en production végétale et leurs applications <sup>2</sup>	
1973	Janzen	Tropical agroecosystems <sup>2</sup>	
1976	INTECOL	Report on an International Programme for analysis of agro-ecosystems <sup>3</sup>	
1978	Gliessman	Memorias del Seminario regional sobre la agricultura agricola tradicional <sup>3</sup>	
1979	Cox and Atkins	Agricultural ecology: an analysis of world food production systems <sup>1</sup>	
1981	Gliessman et al.	The ecological basis for the application of traditional agricultural technology in the	
1083	Altiori	A groecology <sup>1</sup>	
1965	Douglass (ad.)	Agricultural sustainability in a shanging world order <sup>1</sup>	
1904	Arrignon	Agricultural sustainability ill a changing world order	
1907	Computer	Agro-ecologie des zones andes et sub-indinides	
1907	Altiori	A grocelegy. A new recease and development periodic for world agriculture <sup>2</sup>	
1909a	Gliassman (ad.)	Agroecology: A new research and development paradigm for world agriculture <sup>1</sup>	
1990	Caparali	Ecologie por l'agricoltural	
1991	Altiori	Ecologia per l'agricoltura A grace alegue the science of susteinable equivalence $(3rd adition)^{1}$	
1995	Gliassman	Agroecology: the science of sustainable agriculture (51d edition)	
2002		Agroecology, ecological processes in sustainable agriculture	
2003	Dalgaard et al.	Agroecology, scaling and interdisciplinarity <sup>2</sup>	
2003	Francis et al.	Agroecology: the ecology of food systems	
2004	Clements and	New dimensions in agroecology.	
2007	Shrestha (eds.)		
2007	Gliessman	Agroecology: the ecology of sustainable food systems	
2007a	Warner	Agroecology in action: extending alternative agriculture through social networks <sup>1</sup>	

<sup>1</sup> Book.

<sup>2</sup> Journal article.

<sup>3</sup> Conference proceedings or report.

title, that in effect applied the meanings of agroecology that predominated in this period. The German zoologist Friederichs (1930), who also worked in the tropics, published a book on agricultural zoology and related ecological/environmental factors for plant protection. This book presented different pest management strategies, including biological control and the role of natural habitats for pest management, and evaluated the economic impact of pest damage. His approach was very similar to that of Tischler. A second important book on agroecology was published by the U.S. agronomist Klages (1942), whose article in 1928 (Klages, 1928) may be one of the first papers dealing with agroecology without explicitly using the term, and this included research on the distribution of crop plants using a physiological basis. He also analysed the ecological, technological, socioeconomic and historical factors influencing their production; his vision is quite different from that of the zoologists. Although Klages (1942) used the term agroecology only once, his contribution and that of Friederichs (1930) can be seen as the basis for later publications about agroecology. Thus the first scientists to introduce agroecology were rooted in the biological sciences, particularly zoology (Friederichs, 1930) and agronomy and crop physiology (Klages, 1928, 1942; Bensin, 1928, 1935). At the end of the 1960s, the French agronomist Hénin (1967) defined agronomy as being 'an applied ecology to plant production and agricultural land management'. This is not far from Bensin's definition, without actually using the word 'agroecology'. The Italian scientist Azzi (1956) defined 'agricultural ecology' as the study of the physical characteristics of environment, climate and soil, in relation to the development of agricultural plants, e.g. the quantity and quality of yield and seeds. However, he did not include entomological aspects in his analysis. The foundation of his work was already laid 30 years before (Azzi, 1928, 1942).

## 2.2 Expansion of Agroecology: 1970s-2000s

From the 1970s agroecology continued to be defined as a scientific discipline, but also gradually emerged both as a movement and as a set of practices beginning in the 1980s (Fig. 1). Here the general trends toward movements and practical applications are described, with specifics discussed in the 'country' examples since they are generally case-specific. We also observe the close association today between focus on agroecology and work in sustainable agriculture, often by the same people in science and development.

Concerning agroecology as a scientific discipline, greater historical detail from the 1970s until present is given in Hecht (1995), Francis et al. (2003) and Gliessman (2007). They noted that through the 1960s and 1970s there was a gradual increase in apply-

ing ecology to agriculture, partially in response to the Green Revolution that created greater intensification and specialisation. During this period an important influence also derived from research on traditional farming systems in tropical and subtropical developing countries (e.g. Janzen, 1973). Organic farming as an alternative model was discussed in relation to agroecology, for example by Rosset and Altieri (1997) and Guthman (2000). This period was marked by an increasing interest of an ecological point of view on agriculture. The key concept of agroecosystems emerged in the 1970s. It was suggested by the ecologist Odum (1969, quoted in Altieri 1995), who considered them as 'domesticated ecosystems', intermediate between natural and fabricated ecosystems. Since the beginning of the 1980s, agroecology has emerged as a distinct conceptual framework with holistic methods for the study of agroecosystems. Agroecology became defined as a way to protect natural resources, with guidelines to design and manage sustainable agroecosystems (Altieri, 1989a; Gliessman, 1997). Conway (1987) further developed the concept and identified four main properties of agroecosystems: productivity, stability, sustainability and equity. As its influence grew, agroecology contributed to the concept of sustainability in agriculture, mainly applied at the level of the farming system and supported in the proceedings of a conference edited by Douglass (1984), and later expanded by Gliessman (1990) and Altieri (1995). During the 1990s, agroecological research approaches emerged, several textbooks were published, and academic research and education programmes were put into motion, in particular in the USA. Recently, higher education programmes in agroecology have been developed in the USA and Europe.



Fig. 1 Diversity of current types of meanings of agroecology

Finally, agroecology as a scientific discipline went through a strong change, moving beyond the field or agroecosystem scales towards a larger focus on the whole food system, defined as a global network of food production, distribution and consumption (Gliessman, 2007). In this perspective, producers and consumers are seen as actively connected parts of the system (see also Hill, 1985). This entails a new and larger definition of agroecology as 'the integrative study of the ecology of the entire food systems, encompassing ecological, economic and social dimensions, or more simply the ecology of food systems' (Francis et al., 2003). However, we observe that more restricted definitions of agroecology as a discipline, focussing on the field or agroecosystem scales, are still favoured in different countries as described later.

In general, the environmental movements in the 1960s often emerged as a consequence of the unexpected impacts of industrialised agriculture after the Green Revolution. Researchers with narrow focus on short-term yields and economic returns considered environmental and social factors to be externalities. Public policies rarely considered the environmental impact of agriculture, nor the social consequences of a uni-dimensional rural development focussed on production and economics. This environmentalism was primarily concerned with the impacts of toxic substances, in particular pesticides, on the environment. Other non-agricultural topics of these environmental movements included industrial pollution, nature conservation, and distribution of benefits. Nevertheless, in the 1960s, 1970s and 1980s the term agroecology generally was not used to explicitly describe a movement. This started in the 1990s, especially in the USA and Latin America, when the word started to be used to express a new way to consider agriculture and its relationships with society.

Almost in the same period a third word usage emerged, that of recognising a set of agricultural practices which aims at developing a more "environmentalfriendly" or "sustainable" agriculture. An international example is described in LEISA (2008). One of the origins of agroecology as a practice was laid during the 1980s in Latin America. It was seen as the basis for an agricultural development framework, supported by ecologists, agronomists and ethnobotanists working especially in Mexico and Central America. Agroecology helped local farmers to improve their indigenous farming practices as an alternative to a high input, chemical-intensive agriculture promoted by international corporations (see Altieri, 1989a, 1995; Gliessman, 2007). Practices such as conservation of natural resources, adapted soil fertility management and conservation of agrobiodiversity are the practical basis for the different agroecological movements in Latin America (see country example Brazil). Another example for agroecology as a practice is described by Arrignon (1987), who illustrates technical, more adapted methods in agriculture such as water and livestock management or anti-erosion measures as a basis for rural and sustainable development in arid and subhumid areas. Today there are many different types of movements sharing this view which do not explicitly use the term agroecology, including those of multinational chemical and seed companies that define their new-generation products and transgenic crops as essential to long-term sustainability.

## **3 Examples from Different Countries**

#### 3.1 Case Study of the USA

In the USA, a long history of agroecology as a science began with the work of the agronomist Bensin (1930, 1935), concurrent with research of crop physiologist Hanson (1939) and agronomist Klages (1942). Thus the foundation in the USA was mainly laid in agronomy. According to Hecht (1995), a second advance in agroecology in the 1960s and 1970s was shown by a gradual increase in applying ecology to agriculture. At the same time as the Green Revolution with its non-ecological, chemical-intensive practices, maximum yield breeding strategies, and monoculture specialisation - there was a reaction that promoted a renewal of agroecology. Since the early 1970s there has been an enormous expansion of literature with an agroecological perspective. For instance, Cox and Atkins (1979) provided a broad overview and in-depth analysis concerning the dynamics of agroecosystems, including political, economic and energyrelated questions. An important influence was injected from research on traditional farming systems in developing countries, especially in Latin America (e.g. Gliessman et al., 1981; Altieri 1989a, for more examples see Hecht, 1995). Researchers recognised that traditional management of agroecosystems in these countries represented ecologically based strategies for agricultural production, often linking crop and animal production as well as natural resources. Later, the scientific focus of agroecology gained influence in U.S. west coast universities, incorporating entomology, agronomy and ethnobotany (Altieri, 1993). Agroecology had matured to look more closely at the potentials of building biological connections in agriculture.

In the first half of the 20th century, biological interactions were examined as part of the emergence of ecology as a science, but they were strongly neglected from the 1950s to the 1970s due to the large use of pesticides in crop protection. After two to three decades of decline, the evidence of negative pesticide drawbacks revived the study of ecological agriculture that was tied to advanced knowledge in biology. In this period, agroecology contributed to the emergence, definition and consolidation of the concept of sustainable agriculture (Douglass, 1984; Altieri, 1989a, 1995; Gliessman, 1990, 1997). Finally, the initial definitions for agroecology were expanded to "the ecology of food systems" (Francis et al., 2003, Gliessman, 2007).

The environmental movement in the USA appeared in the 1960s (Altieri, 1989a, 1995; Hecht, 1995). Environmentalism was primarily concerned with the impacts of toxic substances, in particular pesticides, on the environment. The publication of Rachel Carson's book 'Silent Spring' was one of the sparks in the early 1960s, providing a critical appraisal of the relationships among agricultural technology, science and nature (Carson, 1964). Today, agroecology plays an important role as a movement towards extending alternative agriculture, through agroecological partnerships between farmers and extension specialists in social networks (Warner, 2007a).

Agroecological partnerships as a movement (Warner, 2007a) suggest that the term 'agroecological practices' was commonly used to describe techniques for improving product quality while avoiding environmental impacts (Warner, 2005, 2007a, b). Other authors identify traditional knowledge systems (e.g. Norgaard, 1984; Raza, 2007) and resource-poor small farmers (e.g. Altieri, 2002) as target groups for agroecological transitions. Warner focuses on the heart of industrial agriculture, where conventional growers employ seasonal foreign workers. He works toward realizing Rachel Carson's 'dream', setting agroecology

in action through socio-technical networks associating producers, extension agents and consumers-citizens who support such alternatives (Warner, 2007a).

In summary agroecology in the USA was first explored by scientists concerned with environmental pollution from agriculture who built up a scientific corpus based on analyses of traditional and conventional practices. As this expanded to larger scale agriculture, the research evolved into a scientific discipline and laid a foundation for agroecological movements in supporting sustainability, rural development, and environmental improvement, all of which helped to promote agroecological practices.

## 3.2 Case Study of Brazil

In contrast to the USA, the foundation of agroecology was laid in Brazil with different types of movements, based on traditional agricultural practices, and not from science. These movements emerged in the 1970s as different forms of alternative agriculture, from a critical appraisal of the effects of agricultural modernisation on farmers and subsequently towards promoting family farms, as well as food sovereignty and autonomy. In the 1970s, the agronomist Lutzenberger (1976) strived to influence environmentalists and farmers' groups, often supported by NGOs and the Catholic Church, towards an alternative agriculture. With his "Brazilian ecological manifesto: 'The end of future?'", Lutzenberger launched a crusade against contamination by pesticides. He subsequently became involved in practising and teaching organic agriculture (Lutzenberger, 1981). This movement grew from concern about environmental deterioration, and also traditional and small farmers' social exclusion from agricultural modernisation (Norgaard, 1984). Lutzenberger later served as national secretary of the environment for Brazil.

Early in the 1980s, both the first "National meeting of alternative agriculture" (convened by the Federation of Brazilian Agronomists) and the creation of the "Advisory body and services to projects in alternative agriculture" (AS-PTA) formalized a network of organisations in 10 Brazilian states (Canuto, 1998). AS-PTA's work led to the first 'National Meeting of Agroecology' in 2001. Among its objectives, this meeting aimed at making agroecology more visible and at lobbying in the national elections. As a result, a Brazilian law in December 2003 gave formal recognition to agroecology under the umbrella of organic farming (Bellon and Abreu, 2006). In this law, participatory guarantee systems (Oliviera and Santos, 2004) and political dimension of agroecology (Byé et al., 2002) were described as important to support small farmers and foster rural communities.

Byé et al. (2002) highlight the appearance of the Ecovida network of agroecology in three states of south Brazil in 1998, as a militant process contributing to a social alternative. For Ecovida, the objectives were to break with organic agriculture third-party certification systems and formal markets, and to approach local markets through a partnership with consumers. Likewise, the Association of Organic Producers from Paraná (AOPA), created in 1995, became the Association for the Development of Agroecology in 2004. The National Articulation of Agroecology (ANA) appeared in 2002 as a space for convergence of movements, networks and organisations from civil society, bringing together groups involved in concrete experiences to promote agroecology and sustainable development in various regions in Brazil.

The Agroecological Movement of Latin-America (MAELA) declared in 1998 its "opposition to degrade nature and society". It advocated "the management and control of natural resources without depending on external inputs (chemicals and genetically modified organisms)" and indicated "its assistance to promote, exchange and extend local experiences of civil resistance and to foster the generation of alternatives to use and maintain local varieties" (MAELA, 2000, quoted by Sevilla Guzmán, 2001).

The Brazilian technical assistance and rural extension public policy also promotes the implementation of agroecological principles through participatory approaches (MDA, 2004). This contributes to the new paradigm 'sustainable rural development', wherein agroecology is considered as one eligible approach (Sevilla Guzmán, 2001, 2002). In public extension services (EMATER) from the southern state of Rio Grande do Sul, agroecology generates a wide disciplinary matrix integrating various forms of knowledge, skills and experiences from distinct social actors, thus giving support to the new paradigm of rural development (Caporal et al., 2006). The authors are extension workers trained in Spain by the sociologist Sevilla Guzmán (2001, 2002). However, they also differentiate "agroecology" from a specific farming type, such as a production system or an agricultural technology (Caporal and Costabeber, 2000), and focus the term on alternative or sustainable agriculture for smallholders. To support their arguments and experiences, they refer to scholars such as Altieri (1989b) to optimise agroecosystems as a whole and not one single production system or activity. They also refer to Gliessman (2007) in arguing that when converting to ecologically based management, the transition levels cannot be attached to any one specific farming situation.

In general, social movements based on sovereignty and autonomy of the local populations are considered as very important and found as a common outlook in Latin America (e.g. see Caporal and Costabeber, 2000). Today in Brazil, this is stronger than ever in the southern federal states of Rio Grande do Sul, Paraná (Caporal and Morales Hernandez, 2004), and Santa Catarina where agroecology was recently institutionalised. This institutionalisation of agroecology (Brandenburg, 2002) is also criticized. For Abramovay (2007), agroecology cannot be the official doctrine at state level, since such a position would counteract scientific progress or impede the development of other production regimes, namely those that tend to serve several objectives in a given institutional context.

On the scientific side, the Brazilian Association of Agroecology (ABA) was created in 2004 (Zonin, 2007). Agroecology was officially recognised recently as a science by the Brazilian Agricultural Research Corporation (EMBRAPA, 2006). For them, agroecology is the ecology of food systems, following Francis et al. (2003), and historically emerging from alternative agriculture and small family farms. The research dimensions integrate renewed conceptual bases and methods, grounded in the work of Altieri (1995) and Gliessman (1997). Emphasis is on agroecosystems and agrobiodiversity in family farms, using systemic, interdisciplinary and participatory approaches, and also to better integrate indigenous knowledge. Agroecology is seen both as an emerging science and as a field of transdisciplinary knowledge, influenced by social, agrarian and natural sciences, especially applied ecology. However EMBRAPA also clearly stated that agroecology should be a foundation to promote sustainable agriculture and rural development (EMBRAPA, 2006). This was strongly initiated through education programs promoted by Altieri and Sevilla Guzman.

As EMBRAPA now recognises agroecology both as a scientific perspective and as a social movement, research workers are re-connected with family farmers' situations, but also with extension workers and high level agricultural training curricula. A final example for a combined scientific and movement approach for agroecology is the Latin American Scientific Society of Agroecology (SOCLA, 2007). Its goal is to promote the development of agroecology as the scientific basis of a sustainable development strategy in Latin America which emphasises food sovereignty, conservation of natural resources and agrobiodiversity and empowers rural social movements.

In summary, agroecology in Brazil was first a movement for rural development and environmental aspects in agriculture. This stimulated a search for alternative practices, now more often called agroecological practices, which were also related to the dynamics of organic farming. In recent years, agroecology has also been considered a scientific discipline, based on an adaptation of the U.S. scientific interpretation of agroecology with an integration of social dimensions.

#### 3.3 Case Study of Germany

Germany has a long history of the scientific discipline of agroecology. From 1930 to present, most research on agroecology has been located within the different faculties of agricultural science and has provided much of the major foundation for agroecology as a science. The starting point was applied zoology with questions concerning crop protection (Friederichs, 1930), and later the ecology of agricultural landscapes (Tischler, 1950, 1953, 1959, 1961, 1965; Heydemann, 1953). Brauns (1985) further developed and broadened agroecology by analysing industrialisation of agriculture and environmental impacts such as herbicides, fertilisers, and water pollution on agroecosystems and their agro-biocoenosis. Agroecological research was extended to the Tropics and Subtropics by Koch et al. (1990). They described the need to thoroughly analyse site parameters, and biocoenosis in agroecosystem, to be able to elaborate strategies and management options for more sustainable local cropping systems.

In addition to the classical ecological factors within an agroecosystem, Schilke (1992) presented, the economic and political influences on agriculture, as well as social consequences for the rural population. Although Schilke does not provide any definition of agroecology, and although his book is not a classical scientific book but rather was written for high school students, he considers this field as the ecology of the food systems (as in Francis et al., 2003). The most recent book by Martin and Sauerborn (2006) finally combines basic ecology and applied agronomy such as farming, crop production and crop protection.

In Germany, a still more restrictive use of the term agroecology in relation to scale is commonly used, and this can be described as agroecosystems ecology or ecology of the agricultural landscapes, or even restricted to the field scale (Fig. 1). For example, Martin and Sauerborn (2006) described agroecology as the science of the conditions for the existence of organisms in the environment, which is managed by man toward the production of certain crops. This is in agreement with the tradition of Tischler (1965), who defined agroecology as the science of life events in the agricultural parts of the landscape. The definition of agroecology provided by the Department of Crop Science (section of agroecology) at the University of Göttingen (2008), probably summarises best both the evolution and the most common current definition of agroecology in Germany: 'Agroecological analyses focus on plant and animal communities, food web interactions, and conservation biology in temperate as well as tropical agricultural landscapes and agroecosystems'. This meaning is widely used in Germany, as compared to the USA where a broader definition including food systems tends to overshadow the narrow field or landscape meaning.

Thus far, the term agroecology as related to movements is more or less nonexistent in Germany. Normally the terms 'environmental movement' or 'ecological movement' are used. These started in the 1970s in opposition against different types of environmental pollution from industry or from nuclear power stations (Brüggemeier and Engels, 2005). In the 1980s the objectives of these movements expanded to include topics such as nature conservation, death of forests in Germany due to acid rain, destruction of tropical forests, or destruction of the ozone layer. Thus far, in only very rare cases some NGOs (e.g. AGRECOL, 2008) speak of an agroecological movement; yet these NGOs are working almost exclusively in Latin America. To summarise for Germany, agroecology is almost exclusively considered within the scientific sphere with a relatively similar interpretation today as used in the past. The focus ranges from field to landscape analyses, mainly based on ecological and biological scientific approaches.

## 3.4 Case Study of France

In France, agroecology is not established as a specific scientific discipline. The French Institute for Agricultural Research (INRA), the largest research organization in agriculture in Europe, has not yet positioned itself nor provided a definition of agroecology, even if the word is becoming more broadly used. At present, the term is used with meaning not far from that of Altieri (1995), focussing on the analysis of agroecosystems for the design of agricultural systems. Agroecology is mentioned only once in the general conclusions of a recent French book on agronomy (Doré et al., 2006). This rare use of the word agroecology in science seems to be strongly related to two historical trends. The first is that agronomy until recently was clearly separated in education and research institutions from the scientific disciplines dealing with crop protection and breeding, and to some extent from ecology. This did not favour integration with agronomy, in particular knowledge about the biotic components of the agrarian system, which have only been integrated in recent years. The second factor is the long history of the discipline of agronomy in France (Robin et al., 2007). Hénin (1967) has given the already mentioned definition of agronomy, and Sebillotte (1974) gave scientific ground to the discipline, agronomy in France has emphasized a systemic approach to agriculture. This led to the development of two scientific directions in agronomy: (i) an analytical direction towards the ecophysiology of plant species, and (ii) a more holistic direction considering the entire agroecosystem, including attention to farming techniques and economic and environmental performances as well. Although the cultivated field, or the plot, was considered as the primary level of attention, other larger scales such as the farm level and the association among farm units and the rural territory also became relevant to agronomy. This includes issues such as maintenance of soil

fertility, quality of product at harvest, and the environmental consequences of agricultural practices such as pollution from nitrate or phosphate. Through these changes, agronomy in France has become enriched with knowledge and concepts from other disciplines, especially from the social sciences, but yet has not changed its name thus far. As a consequence, there were several similarities in regard to conceptual work between U.S. agroecology (sense of Altieri, 1995) and agronomy in France: holistic approach, integration of non-production dimensions and including the social dimension such as analysis of farmers' attitudes and practices. These are among the trends within agronomy in France, which have lasted for more than three decades. But it is clear that what is considered by others as characteristics of agroecology (sense of Francis et al. (2003) and Gliessman (2007), and particularly the ecology of the food system remains unrecognized in France.

Nevertheless the term 'agroecology' is more and more used in France. Interestingly, the English, German and Spanish web pages of Wikipedia (2008) present agroecology as a science, whereas the French web pages define agroecology as a practice, and a certain type of agriculture, which does not only respect the various ecosystems, but integrates the economic and social dimensions of human life. In fact, in many cases the term agroecology is used for describing, in a sometimes very vague manner, a way to produce more ecologically sound agricultural products in the field, or at the farm level. Sometimes agroecology is even used as a synonym for organic or ecological farming. A practice-oriented approach to agroecology has emerged over the past two decades. Although Arrignon (1987) did not give a precise definition for agroecology, he described technical, more adapted methods in agriculture as a basis for rural development in arid and sub-humid areas. This has been expanded by various institutions, with a definition of agroecology as an approach to integrate more ecological aspects into agriculture, and with special emphasis on the field scale, e.g. soil fertility conservation, water management, closed and improved nutrient cycling. Different French research institutions such as CIRAD (2007), which focuses on tropical and sub-tropical agriculture or the Non-Governmental Organisation Agronomes & Vétérinaires Sans Frontières (2007) have implemented this concept within their research and technical

programmes. The latter does not give a more precise definition of agroecology than "an integrated approach of crop production, animal production and the environment". For CIRAD on the contrary, agroecology has been defined far more precisely as a technology which brings agriculture closer to the wild ecosystems, and uses the term ecological engineering (Capillon, 2006). This context is mainly based on conservation agriculture with no-tillage and mulching. Recently, the more plot-oriented approach has been expanded to include the level of field to farm linkages. It may be worth noting that this set of technologies was first applied by CIRAD outside France, for instance in Brazil, before being applied in Southeast Asia and Africa, and then re-imported to France. This technology context is now incorporated in CIRAD publications, with agroecology as a scientific discipline (sense of Altieri (1995)). In the same sense, but not within a research framework, authors like Rabhi (2007) defined agroecological techniques including recycling organic matter and using natural organisms to control pests and diseases. The "mother earth" and its organisms have to be respected in applying 'agroecological techniques', inspired by natural processes, for agricultural production (Rabhi, 2007). Humanism and solidarism are also important. Also for the editors of Terre and Humanisme (2007), agroecology is more a philosophy of ethics, for it also includes societal aspects.

As in Germany, the environmental movement in France started in the 1970s. Topics varied from environmental pollution from industry and from nuclear power stations. Some concerns did exist about industrialized agriculture, and different agricultural trade-unions or NGOs promoted alternative agriculture systems. But their actions were hardly realised in broader environmental movements until recent years. If one can presently speak of an agroecological movement in France, it is more in the sense to promote organic farming or more ecological friendly agriculture.

To sum up for France, agroecology was first seen as an alternative way to practice agriculture. In parallel, the evolution of the French scientific discipline agronomy gradually incorporated parts of what is called agroecology in other countries. In recent times, agroecology as a scientific discipline is on the way to becoming established with a similar interpretation as in Germany.

#### 4 Discussion

# 4.1 Today's Scientific Discipline of Agroecology: A Variation in Definitions and Scales

An interesting aspect in the different concepts and the resulting research in agroecology is the range of different spatial scales as well as the mix of disciplines that have been employed over the past 80 years. The many different definitions and descriptions shown in the publications in Table 1 demonstrate how agroecology has changed from focus on the plot or field scale (1930s to 1960s), to the farm, to landscape agroecosystems, and to farming and food systems scales from the 1970s into the 2000s (Fig. 2). Yet the plot and field scale approach persists up to present, with a narrow definition of applying ecological principles to farming practices by some researchers. Currently, the definitions of agroecology given by Francis et al. (2003) and Gliessman (2007) go beyond the concrete spatial scale of field and farm and expand into the full dimensions of the food system. This dimension requires multi-scale and transdisciplinary approaches and methods, to include the study of food production systems, processing and marketing, economic and political decisions, and consumer habits in society. None of these can be confined nor attributed directly to a certain level of scale, but all are connected intimately with each other across scales and through time in different and complex ways.

Although agroecology as a science evolved significantly and definitions have been articulated, a large diversity still is found in approaches and definitions in different countries and regions of the world. One of the broadest definitions was provided by Francis et al. (2003) with agroecology as "the integrative study of the ecology of the entire food systems, encompassing ecological, economic and social dimensions", or more simply "the ecology of food systems". This approach possesses a degree of originality and inclusiveness, however clear concepts or new analysis models are not yet defined which combine the different dimensions covered by this broad umbrella for various types of analyses. A second definition, integrating the food system, is provided by Gliessman (2007) with "the science of applying ecological concepts and



Fig. 2 Temporal changes in scale and dimension in the definition of agroecology and in applied research

principles to the design and management of sustainable food systems". This definition clearly emphasises the practical application. The definitions of Francis et al. and Gliessman are in part based on prior definitions and descriptions from Altieri (1989a, 1995). Ruiz-Rosado (2006) calls agroecology a transdiscipline because of its systems thinking and systems approach, using methods and advances from various disciplines and taking into account local knowledge where ecological, social and economic concepts and principles are applied in a reasonable manner. Similarly, Buttel (2007) describes agroecology as an interdiscipline that includes the social and human sciences as well as the ecological and agricultural sciences. Dalgaard et al. (2003) defined agroecology as "the study of the interactions between plants, animals, humans and the environment within agricultural systems" in covering "integrative studies within agronomy, ecology, sociology and economics", From their analysis of the situation and state of the art, they concluded that agroecology could clearly be considered as a scientific discipline. A common point in all these approaches is that if someone wants to practice this new discipline - or interdiscipline, or transdiscipline - its operational tools and concepts are still under development and difficult to identify.

Another new possible theoretical approach could follow the holon concept of Bland and Bell (2007), where every system at any level of spatial scale is composed of smaller subsystems, and in turn is a component of larger supersystems, even if this is still difficult to translate into reality. Due to the need to tackle the problems of boundary and change, which are evident for all agroecological research questions, Bland and Bell argue that agroecologists need to take into account how intentionalities seek to create holons (an intentional entity) that persist amid the ever-changing ecology of contexts, and how boundaries can be recognized based on how intentionalities draw and act upon them. This concept needs to be further developed and operationalised.

A more restricted approach in agroecology defines the system boundaries as field, farm, and landscape agroecosystems, without taking into consideration interactions with society, politics and economy. This is almost congruent with Conway (1987), although he also considers sustainability and equity as key properties of a system. A definition was presented in the case study of Germany. Among agroecosystems approaches the concepts and methods will vary depending on the definition of an agroecosystem. At the smallest level of scale, agroecological approaches are restricted to the plot or field scale. Here research almost exclusively analyses crop-insect and crop-weed interaction with a particular emphasis on natural processes, as well as impact of pesticides. For others, the farm is seen as equivalent to an agroecosystem, and still others view an agroecosystem at the scale of a local or regional landscape where agriculture is practised. According to each different choice of scale, applied research methods will vary accordingly. Based on our four country examples, it seems that these two more restricted

approaches dominate in France and to a certain degree in Germany, where the mix between science and social movement in agroecology is less pronounced than in Brazil and the USA. One could conclude that in the areas where science and social movements are mixed, science is more value-laden and needs to incorporate social sciences if its goal is to be achieved.

One major constraint needs to be mentioned in evaluating the use of the term agroecology as a scientific discipline. Germany and USA used the term relatively early, thus it was found in many publications thereafter. However, many more publications exist which did not use the term agroecology in either the title or text, but which can clearly be seen as related to what we now accept as agroecology, based on current definitions. In this sense the use of the terms "agricultural ecology", "agrarian ecology", "ecological agriculture", "crop ecology" or "ecological crop geography" must indeed be mentioned. In countries such as France, it is even more difficult to consider all relevant publications, as synonymous key words for agroecology took a long time to enter the literature and it is very difficult to determine when and where they were used. In general, it can be assumed that there are other valuable publications that clearly recognize agroecology as a science, but which could not be considered here because of the difficulty in identifying them.

One unique example from nearly a century ago [first published in 1915] is the novel *Herland* by Charlotte Perkins Gilman (1992) in which a 2000-year-old society has practiced many of the key concepts in agroecology in producing food, maintaining a clean environment, and solving the challenges of population and food equity. Another is the futuristic agrarian society described by Marge Piercy (1976) in *Woman on the Edge of Time*. Both of these novels would never appear in a literature search, since they do not use the term agroecology, nonetheless they represent another way to visualise and study whole systems and their complex interactions.

Is there any concern about the confusion and lack of acceptance of agroecology as a science? All sciences evolve in their contents and definitions, and this evolution should not be considered as a problem. Nevertheless, at present there are multiple definitions, and different objects, concepts, levels of scale, and research methods. Although this can be seen as richness, our experience reveals that this rich diversity is also a source of misunderstanding. So the question persists, "Is agroecology a science?". Dalgaard et al. (2003) provide convincing evidence, but only as applied to their specific and narrow definition of agroecology as confined to practices and farming systems. It would be erroneous to apply their conclusion to agroecology in general, without considering the great differences among the different meanings and definitions when applying the conventional scientific criteria of communalism, universality, disinterestedness, originality and doubt. As a result, one should not automatically assume that "agroecology" is a science without giving a precise meaning to the word.

## 4.2 Agroecology as a Movement or a Practice

What is an agroecological movement? So far, it is not possible to clearly answer this due to a broad variety of these movements as illustrated in the different country cases. An agroecological movement can be a farmers' group working for food security, sovereignty, and autonomy. Or it could be a more political movement of the local population for rural development (Brazil). Or it can be a farmers' group movement for extending alternative agriculture through social partnerships to better respond to ecological and environmental challenges within relatively specialised agricultural production systems as in the USA. These movements are clearly action-oriented, and in general happen in response to higher common goals such as sustainable development and sustainable agriculture.

Much of the early project work that today we could call "an agroecological movement" did not actually use the term at all, and often they were within the framework of larger environmental activities. This causes certain unbalance in the country studies. In the USA, for example, Hecht (1995) mentions that the environmental movements of the 1970s enhanced agroecology as a science. In contrast, such relations between movement and science were never established in Germany, thus knowledge about similar movements is confined to publications using the term agroecology.

In general, agroecological practices are seen as new, modified, or adapted practices or techniques that contribute to a more environmentally friendly, ecological, organic or alternative agriculture. They are used to improve traditional or indigenous agriculture in developing countries. The different practices are appropriate to their related objectives, definitions, and to certification in the case of organic farming. For traditional agriculture, practices are mainly for soil fertility and organic matter management or resource conservation, or techniques for low external input systems. Biological pest management or soil fertility enhancing techniques are major objectives for environmentally friendly, organic or alternative agriculture. Conversely, agroecology challenges the interpretation of organic farming as mere input substitution rather than redesign of the system (Rosset and Altieri, 1997), and also may soften the division made by ecoagriculture between productive and natural areas (Altieri, 2004).

As with science, the lack of precise definition of agroecological movements or practices may be seen as a weakness. Indeed, everyone supports the goal of a more sustainable agriculture, and thus everyone could claim that his or her own movement could be called agroecology. The same type of difficulty is recognised in describing agroecological practices. At present, an agroecological practice is one that is not ecologically harmful, although this may not be scientifically grounded. In some cases, the environmental problems which practices are supposed to solve are not clarified, adding more to the confusion.

## 4.3 Combined Use of the Three Definitions

In many countries there is a combined use of the term "agroecology" as a movement, as a science and as a practice, and in most situations they are strongly intertwined. In Germany, agroecology has a long tradition as a scientific discipline, and the term is not associated with a movement or with practices. In the USA and in Brazil, agroecology is used to describe all three activities, with a predominance toward science in the USA and a stronger movement and/or practice emphasis in Brazil. In France, agroecology was mainly known until recently as a practice. In the countries where the agroecological movements are well established, the idea of practises is strongly connected, or even incorporated, into these movements. Here, they merge for the objective to develop and assist a transition into sustainable agroecosystems (e.g. Wojtkowski,

2002; Gliessman, 2007) and also with other models such as traditional, alternative or organic farming. There is large overlap in use of these several terms.

In this sense agroecology encourages farmers and extensionists to participate in the design of new systems, and also contribute to social movements. This is particularly the case for Brazil, and to certain extent for the USA and France. In these situations, there is often a link between a political vision (the movement), a technological application (the practices) to achieve the goals, and a way to produce the knowledge (the science). A key-point here for the scientists is to assess how these tight connections may influence the science of agroecology, where there will be application to meet a political vision using a set of technological practices. This association raises serious questions for some who have seen science more as an objective activity that is somewhat disconnected from practice. For example, when the science agroecology is defined as the scientific basis of a sustainable development strategy which emphasises food sovereignty, conservation of natural resources and agrobiodiversity and empowers rural social movements, the science itself may appear as an advocacy activity that will be impacted by diverse goals and applications of results. Instead of considering agroecology as a general matrix including the wider range of disciplines (Caporal et al., 2006), collaborations between agricultural, natural and social scientists should help to clarify such embedded interpretations of agroecology. One must ask, of course, whether this connection between the science and the practice is any different from our accepted linkages between research and recommendation, for example studies of fertilizer rates, types of pesticide that are effective, or scheduling of irrigation. Especially in the USA, where land grant university faculty are often involved in both research and extension, there is often a close connection between these activities.

# 4.4 Agroecology, a History of Oppositions?

In analysing the historical evolution of agroecology, either as a science, movement or practice, it becomes evident that different topics, discussions or debates in certain periods seem to have provoked major changes or reactions within agroecology. The common ground in the U.S. evolutions in agroecology was to find a scientific basis for new alternative agricultural systems. But why did this concern evolve to a new scientific discipline in the USA, but not in France, at least not so soon, where the concern was also present? Maybe because the holistic definition of agronomy, and the included concepts in France could welcome part of the changes needed to target this goal, which was not the case in the USA. Therefore a rupture was sooner necessary in the USA, leading to the creation of a new scientific discipline. For the case of Brazil, a clear rupture can also be mentioned. The expansion of different farmers' agricultural movements in the 1980s and 1990s was finally translated into agroecological movements whose common interests have been canalized under the term agroecology. In addition, to better distinguish their practical approach from industrialised agricultural practices, it seems that the use of the term 'agroecological practices' fulfilled this best. In speaking of agroecological practices since the 1990s, a possibility was found to distinguish them clearly from other conventional practices. Thus, a sort of new identity was created for these practices, which considered for instance more ecological and environmental aspects. In contrast to the USA and to Brazil, the different types of movements have not been the starting point for agroecological movements in France and Germany, or did not provoke a clear reaction into agroecological research. This might be explained by the fact that the agricultural problems in Europe interested or concerned only a limited part of the population. In Brazil on the contrary, a larger part of people are involved in agriculture, and the questions of rural poverty and disparities in agricultural land tenure still are important topics. In the case of Germany the lack of social movement around agriculture can also be explained considering that ecology in general, and in particular ecology related questions to agriculture, have been already well established in research, and thus evolved gradually instead of creating new research fields as for example in the USA. As usual in a scientific community, it can also be noticed that ideas attached to agroecology have crossed national borders. Therefore, many authors attached to a specific institution also gained and communicated knowledge from other countries, for instance through exchanges between Europe and the Americas.

# 4.5 Future Questions and Challenges of the Scientific Discipline of Agroecology

Many open questions remain for the scientific discipline of agroecology, especially for those who embrace the definition as the ecology of food systems. What new concepts, new models, and new methods need to be developed or adapted in order to grapple with this expanded definition of agroecology? Do we need new competencies for researchers and educators dealing with this more holistic and systemic approach (Lieblein et al., 2007b)? These authors raise the question of how higher education and learning activities will need to be modified to deal with systems, uncertainty, and complexity? Lieblein et al. (2000, 2007a, b) have proposed experiential learning strategies, but still the essential contents of agroecology courses have to be clarified and their applications defined for different cultures and perhaps for different ecoregions. Most authors also demand that the scientific discipline of agroecology should provide results for practical application in the design and management of sustainable agroecosystems. It is important to work out the crucial interfaces between the scientific discipline of agroecology and the needs of different stakeholders. It is important that particular participatory or on-farm approaches be employed, which also take into account the social, cultural and ethical dimensions. Although these questions cannot be answered quickly nor easily, and although a certain misunderstanding in using the term agroecology will persist, it will be intriguing to observe the evolution of the terms and their applications in this rapidly changing learning and development landscape. Moreover, it will be interesting to see how the word will escape the fatal challenge of being marginalized as too vague, confusing, and ineffective by scientists, farmers, environmentalists and consumers who want to express their ecological concerns in relation to agriculture and to move these concerns into effective action.

## **5** Conclusions

Today there is certain confusion in use of the term "agroecology". We have described the three main definitions of the term: as a scientific discipline, as a movement, and as a practice. Application of the term depends strongly on the historical evolution and epistemology, that provide the foundation, scope and validity of use of the term, in different countries. The most important influencing factors are (i) the existence of strong social or environmental movements, (ii) the existence of different scientific traditions and their evolutions, and (iii) the search for frameworks and concepts to describe new types of practices or movements.

From the historical analysis it became clear that the scientific discipline of agroecology and its scales, dimensions and definitions distinctly evolved from beginnings in the 1930s. We could illustrate that scales, and with them the definitions, expanded spatially over the next 80 years from the plot or field scale to the farm or agroecosystem scale, and finally leaving a concrete spatial scale or place and entering the entire realm of the food system. Today, all these three different scale approaches still exist within agroecology. The preference to any one of these approaches seems to depend in many cases on the historical evolution in different countries, at least in those we could analyse. In spite of the existence of different approaches and definitions, the new views and dimensions brought into agroecology as a scientific discipline will help facilitate the efforts to respond to the actual challenges of agricultural production, because of increasingly applied systems thinking and interdisciplinary research approaches.

Acknowledgements Many thanks to Guillaume Olivier for his contribution to a scientometric analysis, to Lucimar Santiago de Abreu for her inputs in the Brazilian case study, and to Rob Heslop for some English language corrections.

## References

- Abramovay R (2007) Estartégias alternativas para a extensao rurale suas conseqüências para os processos de avaliação, Proceedings XLV congreso da SOBER (Soc. Bras. Economia, Adm. e Sociologia Rural) - "Conhecimentos para a Agricultura do Futuro", Londrina, 22–25 July 2007, 15 p.
- AGRECOL (2008) Networking sites. Information management and networking for ecological agriculture. http://www. agrecol.de/index.php?option=com\_content&task=view&id= 44&Itemid=51 (accessed January 2008).
- Agronomes & Vétérinaires Sans Frontières (2007) Projet Agroécologie – Manakara. http://www.avsf.org/fr/article.php?rub\_ id=110\&art\_id=194 (accessed June 2007).
- Altieri M.A. (1989a) Agroecology: A new research and development paradigm for world agriculture, Agr. Ecosyst. Environ. 27, 37–46.

- Altieri M.A. (1989b) El "estado del arte" de la agroécologia y su contribución al desarrollo rural en América Latina, in: Cadenas Marin A. (Ed.), Agricultura y desarrollo sostenible. Madrid: MAPA (Serie Estudios), pp. 151–203.
- Altieri M.A. (1993) Ethnosience and biodiversity: key elements in the design and of sustainable pest management systems for small farmers in developing countries, Agr. Ecosyst. Environ. 46, 257–272.
- Altieri M.A. (1995) Agroecology: the science of sustainable agriculture, Westview Press, Boulder, CO, USA, 433 p.
- Altieri M.A. (2002) Agroecology: the science of natural resource management for poor farmers in marginal environments, Agr. Ecosyst. Environ. 93, 1–24.
- Altieri M.A. (2004) Agroecology versus Ecoagriculture: balancing food production and biodiversity conservation in the midst of social inequity. (http://www.wildfarmalliance.org/ resources/ECOAG.pdf).
- Arrignon J. (1987) Agro-écologie des zones arides et subhumides, in: G.-P. Masonneuve & Larose et ACCT (Eds.), Paris, France, 283 p.
- Azzi G. (1928) Agricultural ecology (in Italian), Edition Tipografia Editrice Torinese, Turin, 237 p.
- Azzi G. (1942) Agricultural ecology (in Italian), Edition Dante Aleghieri, Città di Castello, 204 p.
- Azzi G. (1956) Agricultural ecology, Constable & Company, London, 424 p.
- Bellon S., de Abreu L. (2006) Rural social development: smallscale horticulture in Sao Paulo, Brazil, in: Holt G.C., Reed M. (Eds.), Sociological perspectives of organic agriculture: from pioneer to policy, CABI Publishing, pp. 243–259.
- Bensin B.M. (1928) Agroecological characteristics description and classification of the local corn varieties chorotypes. Book, (Publisher unknown so far).
- Bensin B.M. (1930) Possibilities for international cooperation in agroecological investigations, Int. Rev. Agr. Mo. Bull. Agr. Sci. Pract. (Rome) 21, 277–284.
- Bensin B.M. (1935) Agroecological exploration in the Soto La Marina Region, Mexico, Geogr. Rev. 25, 285–297.
- Bland W.L., Bell M.M. (2007) A holon approach to agroecology, Int. J. Agr. Sustain. 5, 280–294.
- Brandenburg A. (2002) Movimento agroecologico: trajetoria, contradições e perspetivas. Revista Desenvolvimento e Meio Ambiente nº 6, July–Dec 2002, Ed. UFPR.
- Brauns A. (1985) Agrarökologie im Spannungsfeld des Umweltschutzes, Agentur Pedersen, Braunschweig, Germany, 395 p.
- Brüggemeier F.-J., Engels J.I. (2005) Natur- und Umweltschutz nach 1945. Konzepte, Konflikte, Kompetenzen, Campus Verlag, Frankfurt, Germany, 379 p.
- Buttel F.H. (2007) Envisioning the future development of farming in the USA: agroecology between extinction and multifunctionality. http://www.agroecology.wisc.edu/downloads/ buttel.pdf (accessed November 2007)
- Byé P., Schmidt V.B., Schmidt W. (2002) Transferência de dispositivos de reconhecimento da agricultura orgânica e apropriação local: uma análise sobre a Rede Ecovida. Desenvolvimiento e Meio Ambiente, Curitiba 6, 81–93.
- Canuto J.C. (1998) Agricultura ecologica en Brasil: perspectivas socioecologicas, Thesis, Universidad de Cordoba, Spain, 200 p.

- Capillon A. (2006) Agriculture durable : faut-il repenser les systèmes de culture ? in: Démeter 2006, Économie et stratégies agricoles, Club Démeter, Paris.
- Caporal F.R., Costabeber J.A. (2000) Agroecologia e desenvolvimento rural sustentável: perspectivas par uma nova Exentsão Rural, Agroecologia e Desenvolviemento Rural Sustentável 1, 16–37.
- Caporal F.R., Morales Hernandez J. (2004) Rio Grande do Sul : vers l'agroécologie, L'Écologiste 5, 40.
- Caporal F.R., Costabeber J.A., Paulus G. (2006) Agroecologia: matriz disciplinar ou novo paradigma para o desenvolvimento rural sustentavel, in: 3rd Congreso Brasileiro de Agroecologia, Florianopolis, Brazil, Anais: CBA, 25 p.
- Caporali F. (1991) Ecologia per l'agricoltura. UTET Libreria, Torino, Italy.
- Carson R. (1964) The silent spring, Fawcett, New York, USA, 304 p.
- CIRAD (Centre de coopération internationale en recherche agronomique pour le développement) (2007) Réseau Agroécologie au CIRAD. http://agroecologie.cirad.fr/index.php? rubrique=dico\&langue=fr#agroecologie (edited January 2007).
- Conway G.R. (1987) The properties of agroecosystems, Agr. Syst. 24, 95–117.
- Cox G.W., Atkins M.D. (1979) Agricultural ecology: an analysis of world food production systems, W.H. Freeman and Sons, San Francisco, 721 p.
- Dalgaard T., Hutchings N.J., Porter J.R (2003) Agroecology, scaling and interdisciplinarity, Agr. Ecosyst. Environ. 100, 39–51.
- Department of Crop Science, Section Agroecology, University of Göttingen (2008) Introduction. http://wwwuser.gwdg.de/ ~uaoe/Agroecology.html (accessed January 2008).
- Doré T., Le Bail M., Martin P., Ney B., Roger-Estrade J. (Eds.) (2006) L'agronomie aujourd'hui, Éditions Quae, INRA, Versailles, France, 367 p.
- Douglass G. (Ed.) (1984) Agricultural sustainability in a changing world order, Westview Press, Boulder, Colorado, USA, 282 p.
- EMBRAPA (Empresa Brasileira de Pesquisa Agropecuária) (2006) Maraco referencial em agroecologia. Ministério da Agricultura, Pecuária e Abastecimento, Brasília, 70 p.
- Editors of Terre & Humanisme (2007) L'agroécologie, vers une révolution agriculturelle, Journal de l'Association Terre & Humanisme 56, 4–11.
- Francis C., Lieblein G., Gliessman S., Breland T.A., Creamer N., Harwood, Salomonsson L., Helenius J., Rickerl D., Salvador R., Wiedenhoeft M., Simmons S., Allen P., Altieri M., Flora C., Poincelot, R. (2003) Agroecology: The ecology of food systems, J. Sustain. Agr. 22, 99–118.
- Friederichs K. (1930) Die Grundfragen und Gesetzmäßigkeiten der land- und forstwirtschaftlichen Zoologie. Vol. 1: Ökologischer Teil, Vol. 2: Wirtschaftlicher Teil. Verlagsbuchhandlung Paul Parey, Berlin, Germany, 417 and 443 p.
- Gilman C.P. (1992) Herland and selected stories. Signet Classics, Penguin Putnam, Middlesex, England, UK.
- Gliessman S.R. (Ed.) (1990) Agroecology: researching the ecological basis for sustainable agriculture, Ecological Studies Series No. 78, Springer, New York.
- Gliessman S.R. (1997) Agroecology: Ecological Processes in Sustainable Agriculture, CRC Press, 384 p.

- Gliessman S.R. (2007) Agroecology: the ecology of sustainable food systems, CRC Press, Taylor & Francis, New York, USA, 384 p.
- Gliessman S.R., Garcia-Espinosa R., Amador M. (1981) The ecological basis for the application of traditional agricultural technology in the management of tropical agroecosystems, Agro-ecosystems 7, 173–185.
- Guthman J. (2000) An agro-ecological assessment of grower practices in California, Agr. Human Values 17, 257–266.
- Hanson H.C. (1939) Ecology in agriculture, Ecology 20, 111–117.
- Hecht S.B. (1995) The evolution of agroecological thought, in: Altieri M.A. (Ed.), Agroecology: the science of sustainable agriculture, Westview Press, Boulder, CO, USA, pp. 1–19.
- Hénin S. (1967) Les acquisitions techniques en production végétale et leurs applications, Économie Rurale, SFER, Paris, France, pp. 31–44.
- Heydemann B. (1953) Agrarökologische Problematik, dargetan an Untersuchungen über die Tierwelt der Bodenoberfläche der Kulturen, Ph.D. thesis, University of Kiel, Germany.
- Hill S.B. (1985) Redesigning the food system for sustainability, Alternatives 12, 32–36.
- Janzen D.H. (1973) Tropical agroecosystems, Science 182, 1212–1219.
- Klages K.H.W. (1928) Crop ecology and ecological crop geography in the agronomic curriculum, J. Am. Soc. Agron. 10, 336–353.
- Klages K.H.W. (1942) Ecological crop geography, Macmillan Company, New York, 615 p.
- Koch W., Sauerborn J., Kunisch M., Pülschen L. (1990) Gedanken zur Agrarökologie in den Tropen und Subtropen, in: Koch W., Sauerborn J., Kunisch M., Pülschen (Eds.), Agrarökologie und Pflanzenschutz in den Tropen und Subtropen, PLITS 8 (2): 109–130, University of Hohenheim, Germany.
- LEISA (2008) Network of magazines on low external input and sustainable agriculture. http://www.leisa.info (accessed November 2008).
- Lieblein G., Francis C., Barth Eide W., Torjusen H., Solberg S., Salomonsson L., Lund V., Ekblad G., Persson P., Helenius J., Loiva M., Sepannen L., Kahiluoto H., Porter J., Olsen H., Sriskandarajah N., Mikk M., Flora C. (2000) Future Education in Ecological Agriculture and Food Systems: A Student-Faculty Evaluation and Planning Process, J. Sustain. Agr. 16, 49–69.
- Lieblein G., Breland A., Østergard E., Salomonsson L., Francis C. (2007a) Educational perspectives in agroecology: steps on a dual learning ladder toward responsible action. NACTA Journal March 2007, 37–44.
- Lieblein G., Østergaard E., Francis C. (2007b). Becoming an agroecologist through action education, Int. J. Agr. Sustain. 2, 147–153.
- Lutzenberger J. (1976) Fim do futuro? Manifesto Ecológico Brasileiro. Porto Alegre (Rio Grande do Sul). Movimento (L&PM Ed).
- Lutzenberger J. (1981) Fundamentos Ecológicos da Agricultura Biológica. Sociedade de Agronomia do Rio Grande do Sul. Ed. 1° Curso de Agricultura Biológica, Porto Alegre, pp. 52–71.
- Martin K., Sauerborn J. (2006) Agrarökologie, Ulmer, Stuttgart, Germany, UTB 2793, 297 p.

- MDA (2004) Ministério do Desenvolvimento Agrário, Secretaria de Agricultura Familiar, Grupo de Trabalho Ater. "Política Nacional de Assistência Técnica e Extensão Rural: Versão Final: 25/05/2004", Brasilia, 12 p.
- Norgaard R.B. (1984) Traditional agricultural knowledge: past performance, future prospects, and institutional implications, Am. J. Agr. Econ. 66, 875–878.
- Oliveira D de, Santos LCR dos (2004) Caderno de formação: certificação participativa de produtos ecológicos. Rede Ecovida de Agroecologia (Fioranopolis, Santa Catarina), 48 p.
- Piercy M. (1976) Woman on the edge of time. Fawcett Books, New York.
- Rabhi P. (2007) Agroécologie. http://www.pierrerabhi.org/blog/ index.php?category/Agroecologie (accessed July 2007).
- Raza S.A (2007) Forces of globalisation, dynamism and diversity in the agro-ecological production system: a study in sustainable development from a hill tribe in Western Himalayas, Int. J. Agr. Res. Gov. Ecol. 6, 666–678.
- Robin P., Aeschlimann J.P., Feller C. (Eds.) (2007) Histoire et agronomie: entre ruptures et durée, IRD Éditions, Institut de Recherche pour le Développement, Paris, 512 p.
- Rosset P., Altieri M. (1997) Agroecology versus input substitution: A fundamental contradiction of sustainable agriculture, Soc. Nat. Res. 10, 283–295.
- Ruiz-Rosado O. (2006) Agroecología: una disciplina que tienda a la transdisciplina, Interciencia 31, 140–145.
- Schilke K. (ed) (1992) Agrarökologie, Metzler Schulbuchverlag, Hannover, Germany, 264 p.
- Sebillotte M. (1974) Agronomie et agriculture, analyse des tâches de l'agronome, Cahier ORSTOM, série biologie 24, 3–25.
- Sevilla Guzmán E. (2001) Agroecologia y desarrollo. rural sustentable: una propuesta desde Latino América, in: Sarandon S. (Ed.), Agroecolog*i*a. El camino hacia una agricultura sustentable, Ediciones Cientificas Americanas, Buenos Aires, Argentina, 25 p.

- Sevilla Guzmán E. (2002) A perspectiva sociológica em Agroecologia: uma sistematicação de sus méthodos et téchnicas, Agroecol. e Desenvolv. Rural Sustent. 3, 18–28.
- Sociedad Cientifica LatinoAmericana de Agroecologia (SOCLA) (2007) Mision y objetivos de SOCLA. http:// agroecologia.pbwiki.com/ (accessed September 2007).
- Tischler W. (1950) Ergebnisse und Probleme der Agrarökologie, Schrift. Landwirtschaft. Fakultät Kiel 3, 71–82.
- Tischler W. (1953) Neue Ergebnisse agrarökologischer Forschung und ihre Bedeutung für den Pflanzenschutz, Mitteilung. Biol. Zentralanst. 75, 7–11.
- Tischler W. (1959) Stand und Möglichkeiten agrarökologischer Forschung, Naturwissenschaft. Rundschau 12, 291–295.
- Tischler W. (1961) Pflanzenschutz in Nordwestdeutschland aus agrarökologischer Sicht, Schrift. Landwirtschaft. Fakultät Kiel 28, 55–70.
- Tischler W. (1965) Agrarökologie. Gustav Fischer Verlag, Jena, Germany, 499 p.
- Warner K.D. (2005) Extending agroecology: grower participation in partnerships is key social learning, Renew. Agr. Food Syst. 21, 84–94.
- Warner K.D. (2007a) Agroecology in action: extending alternative agriculture through social networks, MIT Press, USA, 273 p.
- Warner K.D. (2007b) The quality of sustainability: agroecological partnerships and the geographic branding of California wine grapes, J. Rural Stud. 23, 142–155.
- Wikipedia (2008) Agroécologie. http://fr.wikipedia.org/wiki/ Agroecologie (accessed November 2008).
- Wojtkowski P.A. (2002) Agroecological perspectives in agronomy, forestry and agroforestry, Science Publishers, Enfield, USA, 356 pp.
- Zonin W.G. (2007) Transição agroecológica: modalidades e estágios na RMC (Curitiba Metropolitan Region), Tese de Doutorado. Curso de Doutorado em Meio Ambiente e Desenvolvimento, Universidade Federal do Paraná, 282 p.

# Adaptiveness to Enhance the Sustainability of Farming Systems

Ika Darnhofer, Stéphane Bellon, Benoît Dedieu, and Rebecka Milestad

## Contents

1	<b>Introduction</b>				
2	Three Approaches in Agronomy and Farm				
	Management 4				
	2.1	Improving Crop and Animal Production Based			
		on an Engineering Approach	47		
	2.2	Applying Systems Thinking to Farming	48		
	2.3	An Adaptive Perspective in Understanding			
		Farming Systems	49		
3	Strengthening the Adaptive Capacity of Farming				
	Systems				
	3.1	Learning Through Experimenting			
		and Monitoring	50		
	3.2	Flexibility to Increase Response Options	51		
	3.3	Diversity to Cope with Variability	53		
4	Conc	lusion	54		
<b>References</b>					

I. Darnhofer (⊠)

Abstract During the last decade the context in which farmers must manage their farm has changed rapidly, and often with little warning. Dramatic price swings for agricultural commodities, more stringent quality requirements, new environmental regulations, the debates surrounding genetically modified crops, extreme climatic events, the demand for energy crops, the revision of the Common Agricultural Policy and the consequences of the financial crisis all create uncertainty regarding future threats and potentials. During such turbulent times, a one-sided focus on efficient production is no longer enough. Farmers also need to be able to cope with unexpected events and to adapt to new developments. Based on a literature review, we identify three strategies that strengthen the adaptive capacity of a farm: learning through experimenting and monitoring its outcomes, ensuring a flexible farm organisation to increase the options for new activities by the farm family, and diversifying to spread risks and create buffers. Implementing these strategies enlarges the farmer's room to manoeuvre and allows identifying transition options. These options do not depend only on the farm itself, but also on the farmer's ability to mobilise external resources and to engage in collective action. Change is then no longer seen as a disturbance, but as a trigger for the reorganisation of resources, and for the renewal of the farm organisation and activities. Implementing these strategies comes at a cost, so that farmers need to tackle the inevitable trade-offs between efficiency and adaptability. However, unless farmers master this challenge they may not be able to ensure the sustainability of their farms.

Dept. of Economic and Social Sciences, Univ. of Natural Resources and Applied Life Sciences Vienna, Feistmantelstr. 4, 1180 Vienna, Austria e-mail: ika.darnhofer@boku.ac.at

**Keywords** Adaptive farm management • Adaptive capacity • Evolutionary approach • Complex adaptive systems

## 1 Introduction

Farmers are faced with increasingly stringent environmental regulations, higher quality standards, detailed animal welfare demands and volatile markets, as well as uncertainty regarding the next reforms of the Common Agricultural Policy. Not only is the number of the demands made on farmers increasing, they are also becoming less predictable. The sources of uncertainty include the increasing frequency of extreme climatic events, the demand for energy crops and the debates surrounding genetically modified crops, as well as public health threats stemming from intensive animal production (e.g., BSE, avian influenza, swine influenza). Thus, although farmers have always had to cope with uncertainty, especially regarding weather patterns and prices, it seems that with globalisation the sources of uncertainty are becoming more diverse and the pace of change is increasing (Urry, 2005). This obviously has an impact on farmers' ability to plan ahead and to manage their farm so as to ensure farm continuity. The rapid pace and the often unforeseeable direction of change may increasingly require farmers to keep their farms flexible to be able to respond to new challenges as they arise.

However, the flexibility and adaptability of a farming system has seldom been the target of research on improving farming practices or designing technical innovations. Most of the developments have focused on increasing productivity, improving product quality, optimising production processes, reducing the environmental impact, minimising costs, or maximising profits. The recommendations derived from research were based on the implicit assumption that the general conditions on a farm and its socio-economic context would remain roughly the same. Offering farmers a stable context was also the goal of government policies which stabilised commodity markets and controlled imports. Although there were changes, these were introduced gradually and in a predictable way.

With liberalisation and globalisation leading to an increasing interconnectedness of markets and sectors,

the assumptions of gradual and predictable change are increasingly challenged. The dynamics of the farming system and its context will thus need to receive more attention than they have in the past. To understand these dynamics better, insights derived from complex adaptive systems can be applied to farming systems. The complex adaptive systems view draws attention to the fact that change can be sudden and dramatic and that the links between components are important, rather than the components themselves (Manson, 2001). Systems are also adaptive, i.e. are in constant co-evolution with their environment (Rammel, 2003).

The aim of this paper is to succinctly review three approaches to farming and to indicate how the concepts from co-evolution and complex adaptive systems can be applied to increase our understanding of the sustainability of farming systems. To clarify how the adaptive approach differs from previous approaches, we first briefly review the theoretical concepts and assumptions underlying the engineering approach and the farming systems approach. We then elaborate on the implication of evolutionary and complexity thinking on the understanding of adaptive farming systems. We believe that including the dynamic dimension of evolving farming systems can make an important contribution to understanding how farming systems can be more sustainable in a rapidly changing world.

## 2 Three Approaches in Agronomy and Farm Management

We distinguish between three broad approaches that co-exist in the farm management discourse: the engineering approach, the farming systems approach and the adaptive management approach. These three are loosely defined and we are aware that we cannot do justice to the wide variety of disciplinary refinements. However, using a broad-brush approach allows highlighting the differences in the underlying assumptions between schools of thought (see Table 1). Although the three approaches were developed consecutively, they can all be found in current research, policy formulation and practice.

Characteristic	Engineering	Farming systems	Complex adaptive systems
Underlying theory	Positivism, reductionism	General systems theory, 'simple systems', system dynamics	Co-evolution, complex systems, adaptive systems
Systems view	Mechanistic systems, Newtonian science	Systemic view: system is made of parts that interact, focus on the parts	Hierarchically nested systems, various temporal and spatial scales, properties of complex adaptive systems: emergence, hysteresis, etc., focus on interaction between parts
Time	Atemporal: time not taken into consideration	Atemporal but some linear projection into the future; no change in the dynamics of a system	Time is a key variable: 'history matters', path dependency, irreversibility
Dynamics considered	Static approach, steady-state, equilibrium view	Static approach, equilibrium view, relationship between elements does not change	Perpetual disequilibrium, non-linear dynamics, adaptability: the dynamics change over time, co-evolution
Context	Irrelevant ('one size fits all'), allows for technological blueprints	Context matters: differences between locations is important, farmer perception needs to be taken into account, focus on agricultural sector	Context is constantly changing, change can be unexpected in strength, timing and direction, due to interactions need to include all sectors, not just agriculture
Inclusion of social sciences	Mostly single discipline- driven, some inclusion of neoclassical economics	Interdisciplinary: inclusion of sociology to address farmer perception, farmer participation, economics include some behavioural notions (e.g., agents are boundedly rational), learning	Interdisciplinary, inclusion of insights from psychology such as mental 'traps' and bias typical to information processing by humans, learning as an on-going and interactive process

**Table 1** Key characteristics of the three broad approaches to researching farms and farming systems

## 2.1 Improving Crop and Animal Production Based on an Engineering Approach

The attributes at the core of this approach are: efficiency, constancy and predictability. A problem such as low crop productivity or a threat from pests is identified, and a technological solution developed. Implementing the solution aims at achieving a predictable outcome such as a specific yield level. This approach to farm management is based on an engineering mindset. The goal is to 'design' a crop or animal production system, so as to turn a variable natural process into one that produces standardised commodities in a reliable, predictable and economically efficient way (e.g., Tsai et al., 1987). To achieve this stability agricultural pests are controlled through pesticides, nutrient competition is reduced through herbicides, natural, multi-species grassland is converted into monoculture, water supply for crops is regulated through irrigation or drainage, and field patterns are reorganised to reduce border effects and increase labour productivity (e.g., Fogel et al., 1974; Gotsch and Rieder, 1990). Recent developments, such as robotics and precision farming, aim at adjusting crop management steps to account for field variability by using technological means such as satellite navigation, sensors, computer models and information technology (e.g., Auernhammer, 2001; Bennis et al., 2008; Slaughter et al., 2008). The goal is to control processes, to reduce the range of natural variation of the farming system, and to stabilise the output of the farm so as to ensure an efficient and stable supply of goods and services (Grumbine, 1994; Holling and Meffe, 1996; Okey, 1996).

The solution to a problem is seen as being direct, i.e. there is a clear relation between cause and effect. It is seen as appropriate, i.e. the system has clearly defined boundaries and changes to the system have no effects outside these boundaries. And it is seen as feasible, i.e. relatively simple, without complex interrelationships. Generally, the focus is on an isolated issue, such as the nutrient supply for a crop (Table 1). This is based on the implicit assumption that there are no side effects on other parts of the farm or other spatial scales, and that the relations shown to be relevant over the short term are also the ones that will be relevant over the long term. The world is thus understood as stable, causal relationships are known, and there are no uncertainties. Farmers are seen as a homogeneous group and as business managers or entrepreneurs, so that farmer decision-making is assumed to comply with the model of economically rational agents (van den Bergh et al., 2000; Gowdy, 2007). Being rational, they will all reach a similar conclusion, which is expected to be comparable to the recommendation developed by scientists.

The normative and prescriptive technological solutions derived from this approach led to the strong increase in productivity in the 1970s and 1980s, both in crop and in animal production. This increase in productivity mainly took place in favourable production environments, i.e. good soils, reliable water supply, and a supportive economic and political framework such as ready access to cheap inputs, government-guaranteed output markets and stabilised prices. These favourable environments led themselves to the implementation of 'technological package solutions' (Norman, 2002).

The example of high yielding crop varieties can be used to illustrate the logic of the engineering approach. These are selected to comply with the DUS criteria (distinctiveness, uniformity, stability). These varieties are bred in centralised breeding facilities, with a few large companies claiming exclusive commercial rights to new plant varieties. The quality criteria are mostly breeder-driven and focus on efficiently producing a standardised quality, as required in industrial food processing. They achieve a high yield by being able to make best use of synthetic fertilisers, and tolerate herbicide application. Their performance is thus dependent on large-scale use of fossil fuel-based inputs and processes that help in limiting environmental variability (Tilman et al., 2002; Boody and DeVore, 2006; Wolfe et al., 2008).

# 2.2 Applying Systems Thinking to Farming

The farming systems approach is generally characterised by an increased sensitivity by technical scientists to the complexity and variability of farmers' production environment, the recognition by economists of the limitations of the rational decision-making approach, the heterogeneity of farmers, and the recognition that it is not sufficient to optimise individual crops or animal production systems, as the farm needs to be understood as one system (Bawden, 1995; Hubert et al., 2000; Norman and Malton, 2000; Norman, 2002).

The importance of taking into consideration the farming context became increasingly evident as farmers, especially in less favoured areas, did not adopt the technological packages developed within an engineering approach. Technical scientists thus recognised that both the bio-physical and the socioeconomic components of a farm need to be considered, highlighting the usefulness of an interdisciplinary, systemic and gender-sensitive analysis (Hart and Pinchinat, 1982; Biggs, 1985; Jiggins and Röling, 1994; Dent et al., 1995) (see Table 1).

There was increased awareness of the social nature of heterogeneity between farms and thus the importance of the farmer's perceptions and goals (e.g. Biggs, 1985; Lev and Campbell, 1987; Norman, 2002; Commandeur, 2006; Brodt et al., 2006; Ondersteijn et al., 2006). Linked to these developments was the acceptance by economists that farmers' behaviour could not be understood only through maximisation of profit (Norton, 1976; Colin and Crawford, 2000), so that concepts such as satisficing behaviour (Simon, 1986) and bounded rationality (Kahneman, 2003) were included in the analysis. When making decisions, farmers and farm households also take into account issues such as long-term preferences, security, lifestyle and quality of life (Brossier et al., 1991; Gafsi and Brossier, 1997). Furthermore, farmers were often involved in the research process, e.g. to better understand their norms, values and decision rules (Bellon et al., 1985; Collinson, 2000).

The farm is no longer seen as a mechanistic sum of (more or less) independent parts (Table 1). Instead, a farm is viewed in its entirety, as a system (Osty, 1978; Béranger and Vissac, 1994; Bawden, 1995), as a change in one part of the farm often affects other aspects of the farm organisation. The farming systems approach thus focuses on the interaction between the parts of the system, and on the identification of improvements that were compatible with the whole farming system (Norman, 2002). For example, the livestock farming system approach proposed by animal scientists considers the farmer, the herd and the resources as one socio-technical system (Gibon et al., 1999). This implies that the interactions between its constitutive elements, as well as the self-regulating properties of biological systems, need to be taken into account (see Puillez et al., 2008).

The implications of this different understanding of agronomy and farming systems can be illustrated in how it shapes the selection of crop varieties. Instead of stabilising the production environment to achieve high yields, varieties are selected for being robust in a wide range of conditions, i.e. aim for yield stability across a wide range of temperature, nutrient and water conditions. Standardisation of quality is not seen as desirable, as farmers and consumers have different preferences, and traditional cuisine requires specific qualities found in landrace varieties (Sánchez et al., 2008). Desirable traits are not reduced to yield and qualities, but include system-wide considerations, such as balancing the nutrient needs in the whole crop rotation or the need for straw for animal housing (Wolfe et al., 2008). To take into account these preferences, breeding often takes place in a decentralised setting, with farmer participation (Sumberg et al., 2003; Bocci and Chable, 2008; Østergård et al., 2009).

# 2.3 An Adaptive Perspective in Understanding Farming Systems

As the long-term environmental and social impact of intensive farming systems became increasingly apparent, the various issues related to ecological, economic and social sustainability came to the fore. This led to more attention given to longer-term effects, and raised the challenge to balance short-term productivity increases with the long-term sustainability of farming systems. It also pointed out that the externalities of on-farm practices need more attention, and that the interaction between the farm and its context needs to be better understood. At a fundamental level, it highlighted the need to account for uncertainty as complex dynamics and interdependencies between sub-systems cannot be fully anticipated (EEA, 2001; Mayumi and Giampietro, 2001; Millennium Ecosystem Assessment, 2005; Campbell, 2008; IAASTD, 2009). With societal and farm dynamics being uncertain, there are bound to be unexpected developments that will require flexible adjustment (Table 1).

There had been early calls for technologies to increase farmers' flexibility, to enable them to cope with changing economic environments (Long, 1984; Chambers, 1991, quoted in Norman, 2002; Lev and Campbell, 1987; Park and Seaton, 1996). Petit (1978, 1981) suggested a theory of adaptive behaviour in farm management, based on the observation that farmers interactively adjust both their objectives and their situations. This was illustrated by studying the patterns of change of farms over long time frames, offering empirical evidence that farm structure, activities and organisation can change substantially in response to on- and off-farm dynamics (Bourgeois and Krychowski, 1981; Levrouw et al., 2007; Cialdella et al., 2009).

Against the multi-dimensional background of socio-economic, political and environmental dynamics, changes and adaptations increasingly seem to be essential elements in any approach towards a sustainable farming system. It is thus necessary to understand the (co-)evolution of a farming system with its environment and how this on-going change is reflected in the internal organisation of the farm and the farmer's goals. Given the uncertainties of future developments, previous concepts that guided research – such as stability, income maximisation, technical finetuning or biological optimisation – need to be balanced with concepts such as adaptability, resilience and flexibility.

The theories that might be used to inform an adaptive perspective of farm management are evolutionary theory and complexity theory. The term 'evolutionary' is used for theories that explain the driving forces requiring the adaptations of the system over time, and the mechanisms through which they operate. Evolutionary theories have mostly been developed in ecology. In the context of farms, these theories can help explain how farms generate and adapt to change, and how these processes are intertwined with what happens both at the level of individual farms and the higher level of markets and the farm's environment in general (see Rathe and Witt, 2001). In an evolutionary framework, continual development and change at the farm level is needed to maintain its 'fitness' relative to the systems it is co-evolving with (Cournut and Dedieu, 2004). This on-going change implies that there is no stable state, no single optimal solution, no 'right' development path that can be defined a priori (Rammel, 2003). The evolutionary perspective requires a system to be adaptable, i.e. to be able to perform well according to unknown future conditions and goals that might change over time (Holling et al., 2002; Smit and Wandel, 2006; Rammel et al., 2007; van den Bergh, 2007; Fauvergue and Tentelier, 2008). The objective of management must include initiating and maintaining a diversity of alternative options so as to increase the chance of finding an adaptive response to unpredictable change (Beinhocker, 2006).

The theory of complex adaptive systems is another theoretical approach that focuses on understanding the implication of on-going change, and which emphasises the unpredictability of change (Ison et al., 1997; Levin, 1998; Manson, 2001; Holling, 2001). Complex adaptive systems are systems that involve many components and agents that interact simultaneously and adapt or learn as they interact (Holland, 2006). The theory of complex adaptive systems has been taken up by some researchers in economics and management sciences (e.g., Anderson, 1999; Meyer et al., 2005; Walsh et al., 2006; Teece, 2007), as well as in other social sciences (Stewart, 2001; Urry, 2005). This integration has proven fruitful as it allows one to understand how firms interact with their environment, how the past influences present behaviour, how the components interact (rather than focusing on the properties of the components), and how function is maintained, even though the components may be replaced (Cilliers, 2005; Trigeorgis, 2005). Within natural resource management, it has led to an approach called adaptive management (Lee, 1999; Westley, 2002; Jacobson et al., 2009). Given the similarities in the challenges faced by farmers and other society actors in the face of a rapidly changing context, it would seem that integrating insights from studies of complex adaptive systems and adaptive management can be useful to increase our understanding of the adaptability, resilience and persistence of farming systems.

# 3 Strengthening the Adaptive Capacity of Farming Systems

Emphasising the adaptive capacity of farming systems is based on the premise that the key to coping with rapid and unforeseeable change is to strengthen the ability to adequately respond to change to sustain longterm survival. However, this needs to be balanced with the ability to take advantage of existing favourable conditions, i.e. to perform under current conditions. The challenge for farm management is thus to balance between long-term adaptability and short-term efficiency (Lev and Campbell, 1987; Giampietro, 1997). Studies of natural resource management indicate that the characteristics allowing a social-ecological system to strengthen its adaptive capacity include the ability of the manager to learn, the flexibility of a system and its diversity.

# 3.1 Learning Through Experimenting and Monitoring

The existence of uncertainty and surprise as well as their unpredictable nature (Funtowicz and Ravetz, 1993; Folke et al., 2003) requires a continuous learning process that attunes to new information by reformulating hypotheses and models, and understanding activity implementation as experiments (Westley, 2002; Hagmann and Chuma, 2002). For example, there could be different interpretations as to the cause of an animal disease such as calf scour, leading to different assumptions on how to best tackle the disease. Whereas one farmer might rely exclusively on the veterinarian to treat the disease when it appears, another will experiment with different preventive measures and adapt processes on her farm (Magne and Cerf, 2009). Indeed, farmers' choices are constrained by their personality, preferences and competences, but also by external structures such as the social norms, technologies and the natural environment. Acknowledging that there are different valid solutions for each problem allows one to see that a farmer might find some solutions more useful than others depending, e.g., on her priorities, farming style and context. Learning is thus not seen as an objective attempt to understand the 'world out there', but as based on a

relational understanding of reality: learning allows for a new perspective of challenges and for perceiving new possibilities.

To increase the number of learning opportunities and to structure them, it is useful to experiment and monitor the outcomes. Experimentation allows a better understanding of current system dynamics; for example, the influence of buffer strips on the insect population and reduction of pest incidence. It also allows widening the repertoire of options in case of changes in the context. For example, a farmer might experiment with mechanical weed suppression to reduce herbicide use or experiment with on-farm processing to see whether it would be compatible with work flow and meet consumer demand (Sumberg et al., 2003). In this framework, quantitative information is often less important than understanding the 'rules of the game' and how these rules change. Unexpected outcomes, active experiments to test hypotheses and monitoring through feedback systems allow farmers to learn about local agroecosystems, about the dynamics of social institutions, and about the potential and limits of various technologies and processes, and thus inform how they actively adapt their farm management.

This approach stresses the role of creativity and imagination, and recognises the crucial role of the farmer for the development of a farm over time. Indeed, what a farm can produce with given resources hinges critically on the conceptions, capabilities and projects of the farmer (Rathe and Witt, 2001; Teece, 2007; Gueringer et al., 2009). The farm is thus no longer seen as a device to exploit economies of scale as a response to technological progress. Rather, farms are interpreted as learning systems whose survival and growth strongly depends on the successful generation and integration of new knowledge.

Learning is not limited to experiments a farmer undertakes to understand a specific aspect of his farming system better. Much learning takes place through discussions with others, and when expanding on local and traditional knowledge (Berkes and Folke, 2002; Sumberg et al., 2003). For example, by building on the experiential knowledge of older farmers, who know which pastures remain productive even during drought, a farmer can broaden his options for action when faced with a similar crisis. But monitoring need not be done on an individual basis. It can involve a range of local stakeholders (Couix and Hubert, 2000). A farm monitoring and study group can provide a collective learning environment, in which ideas are shared and the results of experimentation with delayed lambing, flexible bull finishing and cross-breeding are analysed in detail, thereby confirming or disproving ideas for new practices (Seath and Webby, 2000).

Learning also benefits from combining different types of knowledge, e.g. experiential and experimental knowledge (Scoones and Thompson, 1994), from expanding from knowledge of structure to knowledge of function, from understanding about the dynamics of complex systems, and from understanding the complementarities of different knowledge systems such as scientific and traditional knowledge (Folke et al., 2003). Indeed, local knowledge systems can be based on a different conceptualisation of the world compared to science-based farm management (EEA, 2001; Olsson and Folke, 2001; Macé et al., 2007). It is thus important not to dilute, homogenise or diminish the diversity of knowledge systems, but to nurture diversity (Folke et al., 2003). Farmers' learning can benefit from comparing diverse information sources and perceptions, for example, by discussing new ideas with people belonging to different social groups. Off-farm employment or engagement in community organisations (e.g., church, sports club, hunters, fire brigade) allows access to various information sources, different world views, and different understandings of societal trends, consumer preferences or upcoming changes in agricultural policy. By discussing new ideas with a variety of people and reflecting on their views, a farmer may learn new ways to interprete and explain phenomena and thus discover new options to act (Ison et al., 2000; Ondersteijn et al., 2006).

## 3.2 Flexibility to Increase Response Options

In management sciences, the concept of flexibility is seen as a means to face uncertainty and thus also defined in relation to adaptive capacity (Reix, 1979). Generally, there is a distinction between operational and strategic flexibility. Operational flexibility refers to the ability of a system to implement changes in the short term when facing surprises. Strategic flexibility refers to long-term choices and to the capacity to change the structure, the resources, and the competences of the farm in anticipation of, or to react to, changes in the environment. For example, farm households not only need to ensure that they can flexibly change their daily or weekly work schedule to respond to changing weather patterns (operational flexibility), they also need to be able to develop new on- or off-farm enterprises (strategic flexibility).

Tarondeau (1999) further identified three sources of flexibility in production systems: the products, i.e., their diversity and exchangeability; the processes, i.e. the organisation of work and of the technical systems that allows for several processes; and the input specificity, i.e. whether different sources of inputs can be combined or substituted rather than depending on one specific input. This concept has been used to analyse the adaptive capacity of farming systems (Bellon et al., 2004; Lopez-Ridaura et al., 2005). As the available work force is often limited on farms, work organisation is of particular importance: what tasks need to be done, who can perform the task, is it possible to hire skilled workers, can work flows be adapted to react flexibly, e.g. to changing weather (for a review see Madelrieux and Dedieu, 2008).

For example, livestock farms in pastoral systems may be specialised and have only dairy cows, or they may keep a flexible mix of dairy cows, suckling cows, heifers and oxen. Work organisation and processes may be based on regular routines and a clear differentiation of tasks to be performed by each worker, or they may be flexible and make room for contingencies. Regarding inputs, a study by Gueringer et al. (2009) shows the possibility of coping with changing fodder availability through spatial management of grassland fields and flexible fodder purchase. The farmers differed in their combination of harvesting technique, i.e. grazing, harvest as hay or harvest as silage to be stocked as wrapped bales, depending on various factors such as grass regrowth, labour availability, quality schemes (hay-only milk), spatial location of the fields (size of field and distance between individual fields), and available storage options. Farmers may also increase their flexibility by adapting stocking density and herd composition (dairy cows, suckling cows, calves) to available fodder and labour, as well as market demand (Lemery et al., 2006).

This illustrates the complexity with which farmers are confronted on a daily basis, in an attempt to remain flexible while maintaining the overall coherence of their farm. Farmers have to maintain various sources of flexibility over the short term, e.g. fodder sources over one season, and over the medium term, e.g. type and quality of milk and meat produced, as well as in the long term. As a study of change patterns on 14 farms during a 50-year period has shown, farms implement a wide set of adaptability options by flexibly organising the workload of family members, changing the structure of animal production and using off-farm employment, as well as various forms of cooperation with neighbouring farms (Cialdella et al., 2009).

Indeed, flexibility does not depend only on processes internal to the farm but also on its capacity to enrol external resources. Chia (2008) has called this ability 'relational flexibility', i.e. the ability of a farm to mobilise external resources through collective action. This might take the form of processing or marketing cooperatives to promote a specific quality label, the purchase of machinery by several farmers, or establishing a company to run a common biogas plant. On a smaller scale it also includes mutual help and exchanges. For example, the flexibility of the management of grassland fields can be enhanced, if the farmer not only considers her own fields, but is able to arrange with a neighbouring farm for, e.g., a seasonal exchange of plots or labour (Gueringer et al., 2009).

Emphasising flexibility thus highlights that the processes on a farm, its work organisation and the products and services marketed by a farm at any given time merely represent one of several ways in which it could be using its resources. Indeed, the 'productive opportunities' of the farm, even with an unchanged set of resources, are not objectively given. A farm is essentially a pool of resources that can be used and combined in different ways (Penrose, 1997). This means that for a farm, it matters how the components are linked and the way in which the resources are used, not just the resources themselves. What matters is how flexible the arrangements are, and whether or not the selected technological paths enable reversibility. To ensure flexible rearrangement, modularity, i.e. the ability to combine subroutines, actions or resources in different ways plays an important role. Modularity allows farmers to respond to change by combining a set of available resources. The novel combination of available building blocks is often more flexible than trying to anticipate each possible situation with a distinct strategy or resource (Holland, 2006).

#### 3.3 Diversity to Cope with Variability

Clearly, flexibility is in part linked to diversity, i.e. the ongoing development and management of a portfolio of alternative capabilities, opportunities and relationships (Smit and Trigeorgis, 2006). Managing complex systems and uncertain future developments implies spreading risks and creating buffers, i.e. not putting 'all eggs in one basket'. The evolutionary potential of a farm, its ability to initiate new development trajectories, builds on the diversity of co-existing activities, its repertoire of alternative options and innovative activities (Rammel and van den Bergh, 2003). This diversity has been shown to play an important role in the reorganisation and renewal process following disturbance (Folke et al., 2003) as it enlarges a farmer's room to manoeuvre.

One way to approach diversity is at the wholefarm level. Here, activities of the members of the farm household can be diversified, which includes both onand off-farm activities (Bryden et al., 1992; Lemery et al., 2005; Cialdella et al., 2009). A diversity of resources available on the farm, such as family labour, knowledge, networks, arable land, grassland and buildings can be invested in a range of projects. These projects will in part depend on the diversity of opportunities offered by the context, such as proximity of the farm to a city offering employment opportunities or a demand for fresh local products, or a village dairy manufacturing quality labelled cheese or the city council outsourcing services such as composting, snow ploughing or roadside maintenance to farmers. Thus, even if the focus is on the farm, the context needs to be taken into account, as it shapes the diversity of options available. Of course, this diversity can be enhanced by the farmers, especially if they engage in collective action, e.g. such as establishing a cooperative to process and package their produce (Chia, 2008; Cialdella et al., 2009). Also, diversity is not objectively given, but depends on the creativity of farmers to be innovative and creative. Indeed, different perceptions (Magne and Cerf, 2009) and interpretations of societal trends can make the farm and community more robust.

Unfortunately, strategies to build diversity are not yet well understood (Penrose, 1997), as most research efforts have focused on efficiency and specialisation. More research is needed to understand diversification and pluriactivity, the coordination and interaction between the activities leading, e.g., to challenges in work organisation (Fiorelli et al., 2007; Madelrieux and Dedieu, 2008). Indeed, farmers increasingly complain of too high work pressure and lack of work-free time (weekends, vacations). The challenge in building diversity is thus not only to coordinate labour peaks and to ensure the flexibility necessary to accommodate unpredictable events, but also to ensure quality of life through a satisfactory work-life balance for all members of the farm family.

Another way to approach diversity is at the technical system level. Here, the focus is on the role that diversified resources, production processes and type of products play to secure the system and to allow its evolution. For instance, a study has shown that creating and maintaining a diversity of land resources can play a key role in the management of a dairy farm, especially to reduce the sensitivity of milk production to climatic variations (Andrieu et al., 2008). Similarly, a system that allows for a diversity in crops (Østergård et al., 2009) or herd management (e.g., composition, age, uses, etc.) seems to be more resistant to periodic forage shortages, especially when considered at the herd level and over the long term (Tichit et al., 2004).

Diversity can also be approached at the functional level (Elmqvist et al., 2003; Walker et al., 2006). The focus is then not on diversity to allow various responses to change, but on ensuring that a function on the farm, such as ensuring adequate nutrient supply for crops, can be supplied by different processes, such as synthetic fertilisers, compost or adequate crop rotations. At the farm level, functional diversity means that the farm can perform several functions, i.e. is multifunctional, thus addressing a range of societal demands (Wilson, 2008; OECD, 2009). This approach allows one to avoid the sectoral approach to farming, which tends to focus on food, fibre and fuel production, and widen the perspective to the provision of public goods and services such as ecosystem services, cultural landscape or climate change mitigation.

Maintaining diversity in the current activities as well as maintaining diversity of future options implies that not all resources are used efficiently at any one point in time. In other words, there will be apparently redundant resources that are maintained 'in case' something happens, when they might be useful. Keeping unused buildings, machinery or land involves costs and thus reduces efficiency in the short term. However,
diversity and redundancy are an insurance against uncertainty and surprise, and need to be actively nurtured to allow for reorganisation and renewal. A farm will thus have to pursue parallel strategies by exploiting today's capabilities and at the same time exploring new projects. In stable periods, the focus is likely to be on exploiting current strengths while during turbulent periods it will be more important to assess which of the projects are likely to meet the demands of the new context. For example, whereas until the mid-1990s livestock farmers were able to focus on the production of standardised quality, since the BSE crisis in the late 1990s the quality specifications, documentation and traceability requirements have increased dramatically. A farmer that had a range of marketing channels and had experience with quality production thus adapted more easily to the changes than a farmer that had focused exclusively on mass production of standard quality (Chia, 2008). Diversity thus implies a dynamic pro-

cess involving multiple parallel strategies, which allow one to adjust or switch between various alternative paths as the strategy unfolds.

Putting diversity to use implies that decision rules need to change over time, to adapt to changes in requirements, preferences or context. Farmers are thus faced with the challenge of deciding which rules to change when, and how to change them. Holland (2006) has called them the 'credit assignment problem' and the 'rule discovery problem'. The first problem arises because overt information about performance and efficiency is often irregular and partial. Indeed, why a strategy was successful is hard to establish, since the strategy is the result of a long line of choices extending over space and time. Holland (2006) uses the metaphor of a game of chess, where a winning player has little information about which moves along the way were critical to success. The rule discovery problem arises when it becomes obvious that some of the farmer's decision rules are ineffective. Rules serve as tentative hypotheses about the farmer's environment. As that environment evolves, some rules will be progressively disconfirmed. The question then is: how to select a new, more adequate decision rule. As these challenges show, maintaining an appropriate level of diversity, and making use of that diversity are not trivial management issues.

### 4 Conclusion

Farmers have always had to cope with a certain level of change and unpredictability and thus needed to be flexible and adapt to new circumstances. However, with globalisation, sectors and countries are increasingly interconnected, leading to spill-over effects, so that change is different both in speed and variety. Adaptability is no longer just one factor enhancing competitiveness on the market, it has become a key aspect of farm survival.

Enhancing adaptability goes against the recommendations derived from an engineering approach to farm management, which tend to create simplified, specialised farms. Their impoverished diversity limits their capacity to adapt to societal change. Coping with ubiquitous change also demands broadening the perspective found in farming systems, to integrate the options built through collaborative actions by several stakeholders, as well as emphasising dynamics and adaptability.

Understanding the ability of farms to be adaptive raises the challenge to identify and develop methods to capture the dynamics of a system, and analyse which characteristics strengthen or threaten the ability of farms to adapt. Participatory methods are a promising avenue, as most disciplinary scientific models cannot capture the complexity of relationships or their dynamics, whereas farmers juggle them on a daily basis. Thus, researchers face the dual challenge of developing adequate theories and methods to understand the dynamics of co-evolution, as well as ensuring that their recommendations are relevant to real-world decisionmaking.

Learning to live with change and uncertainty requires a fundamental conceptual shift, from assuming that the world is in a steady state to recognising that unexpected change is the rule. Farms and farming systems undergo constant reorganisation, with phases of more or less fundamental modifications. Through our literature review we have shown that learning, flexibility and diversity, in their various forms, play a key role in the strategies of farm households to cope with change. The goal of these strategies is both to recognise the opportunities offered by change and to implement them by initiating transition processes. As learning, flexibility and diversity require resources, they are costly. Farmers thus face the inevitable tradeoffs between efficiency and adaptability. Ensuring the economic, social and ecological sustainability of farms is a dynamic and complex adaptation process, in which strategies and contexts co-evolve.

Acknowledgements We would like to extend our gratitude to the two anonymous referees and to the editor for their constructive criticism and helpful suggestions, allowing us to improve on the paper's earlier version. We also thank the participants at the IFSA 2008 conference in Clermont Ferrand for providing so many case studies illustrating the adaptability of farms.

### References

- Anderson P. (1999) Complexity theory and organization science, Organization Sci. 10, 216–232.
- Andrieu N., Coléno F., Duru M. (2008) L'organisation du système fourrager source de flexibilité face aux variations climatiques, in: Dedieu B., Chia E., Leclerc B., Moulin Ch., Tichit M. (Eds.), L'élevage en mouvement. Flexibilité et adaptation des exploitations d'herbivores, Ed. Quae, Paris, pp. 97–112.
- Auernhammer H. (2001) Percision farming The environmental challenge, Comput. Electron. Agric. 30, 31–43.
- Bawden R. (1995) On the systems dimension in Farming Systems Research, J. Farming Systems Research-Extension 5, 1–18.
- Beinhocker E.D. (2006) The origin of wealth. Evolution, complexity and the radical remaking of economics, Harvard Business School Press, Boston.
- Bellon S., Guérin G., Léger F. (2004) Anticiper les aléas climatiques en programmant des sécurités, in: Duboeuf J.P. (Ed.), Evolutions of sheep and goat production systems: Future of extensive systems and changes in society, Options méditerranéennes, Série A, n° 61, Ciheam/ FAO/ ICZS/ Cirval, pp. 137–146.
- Bellon S., Mondain-Monval J.F., Pillot D. (1985) Recherchedéveloppement et farming system research: à la quête de l'opérationnalité, Proceedings of the colloquium on 'Caribbean farming systems and alternatives for development', held 9–11 May 1985 in Martinique, Univ. Antilles-Guyane, DAC, pp. 467–486.
- Bennis N., Duplaix J., Enéa G., Haloua M., Youlal H. (2008) Greenhouse climate modelling and robust control, Comput. Electron. Agr. 61, 96–107.
- Béranger C., Vissac B. (1994) A holistic approach to livestock farming systems: theoretical and methodological aspects, in: Gibon A., Flamant J.C. (Eds.), The study of livestock farming systems in a research and development framework, EAAP Publ., 63, pp. 5–17.
- Berkes F., Folke C. (2002) Back to the future: Ecosystem dynamics and local knowledge, in: Gunderson L.H., Holling C.S. (Eds.), Panarchy, Understanding Transformations in Human and Natural Systems, Island Press, Washington DC, pp. 121–146.

- Biggs S.D. (1985) A farming systems approach: Some unanswered questions, Agr. Admin. 18, 1–12.
- Bocci R., Chable V. (2008) Peasant seeds in Europe: Stakes and prospects, Cah. Agric. 17, 216–221.
- Boody G., DeVore B. (2006) Redesigning agriculture, Bio-Science 56, 839–845.
- Bourgeois A., Krychowski T. (1981) L'adaptabilité des exploitations laitières: essai d'appréciation de certaines de ses composantes, à partir de douze cas du Maine-et-Loire, Fourrages 88, 3–38.
- Brodt S., Klonsky K., Tourte L. (2006) Farmer goals and management styles: Implications for advancing biologically based agriculture, Agr. Syst. 89, 90–105.
- Brossier J., Chia E., Marschal E., Petit M. (1991) Gestion de l'exploitation agricole familiale et pratiques des agriculteurs : Vers une nouvelle théorie de la gestion, Revue Canadienne d'Économie Rurale 39, 119–135.
- Bryden J.M., Bell C., Gilliatt J., Hawkins E., MacKinnon N. (1992) Farm household adjustment in Western Europe 1987– 1991, Final report on the research programme on farm structures and pluriactivity, Luxembourg: ECSC-EEC-EAEC.
- Campbell A. (2008) Paddock to plate: Food, farming and Victoria's progress to sustainability, The Future Food and Farm Project Background Paper, Australian Conservation Foundation, Melbourne.
- Chambers R. (1991) Complexity, diversity and competence, Toward sustainable livelihood from farming systems in the 21st century, J. Asian Farming Systems Association 1, 79–89.
- Chia E. (2008) La flexibilité relationnelle : rôle des réseaux, groupements et associations d'éleveurs, in: Dedieu B., Chia E., Leclerc B., Moulin Ch., Tichit M. (Eds.), L'élevage en mouvement, Flexibilité et adaptation des exploitations d'herbivores, Ed. Quae, Paris, pp. 137–144.
- Cialdella N., Dobremez L., Madelrieux S. (2009) Livestock farming systems in urban mountain regions. Differentiated paths to remain in time, Outlook Agr. 38, 127–135.
- Colin J., Crawford E. (2000) Economic perspectives in agricultural systems analysis, Rev. Agric. Econ. 22, 192–216.
- Cilliers P. (2005) Complexity, deconstruction and relativism, Theory, Culture and Society 22, 255–267.
- Collinson M. (2000) A history of farming systems research, Wallingford: CABI Publishing.
- Commandeur M.A. (2006) Diversity of pig farming styles: understanding how it is structured, NJAS-Wageningen J. Life Sci. 54, 111–127.
- Couix N., Hubert B. (2000) Promoting collective learning in a land-use management project: Thirteen years' experience in researcher-technician partnership in the Cévennes, France, in: Cerf M., Gibbon D., Hubert B., Ison R., Jiggins J., Paine M., Proost J., Röling N. (Eds.), Cow up a tree, Learning for change in agriculture – Case studies from industrialised countries, Paris: INRA, pp. 121–140.
- Cournut S., Dedieu B. (2004) A discrete event simulation of flock dynamics: A management application to three lambings in two years, Anim. Res. 53, 383–403.
- Dent J.B., Edwards-Jones G., McGregor M.J. (1995) Simulation of ecological, social and economic factors in agricultural systems, Agr. Syst. 49, 337–351.
- EEA (2001) Late lessons from early warnings: The precautionary principle 1896–2000, Environmental Issue Report No. 22, European Environment Agency, Copenhagen.

- Elmqvist T., Folke C., Nyström M., Peterson G., Bengtsson J., Walker B., Norberg J. (2003) Response diversity, ecosystem change and resilience, Front. Ecol. Environ. 1, 488–494.
- Fauvergue X., Tentelier C. (2008) Flexibilité adaptative : biologie évolutive, théorie des jeux et psychologie, in: Dedieu B., Chia E., Leclerc B., Moulin Ch., Tichit M. (Eds.), L'élevage en mouvement, Flexibilité et adaptation des exploitations d'herbivores, Ed. Quae, Paris, pp. 39–58.
- Fiorelli C., Porcher J., Dedieu B. (2007) Pourquoi faire de l'élevage quand on a un autre travail? Rencontres Recherches Ruminants 14, 389–392.
- Fogel M.M., Duckstein L., Kisiel C.C. (1974) Optimum control of irrigation water application, Automatica 10, 579–586.
- Folke C., Colding J., Berkes F. (2003) Synthesis: Building resilience and adaptive capacity in social-ecological systems, in: Berkes F., Colding J., Folke C. (Eds.), Navitaging social-ecological systems. Building resilience for complexity and change, Cambridge University Press, Cambridge, pp. 352–387.
- Funtowicz S., Ravetz J. (1993) Science for the post-normal age, Futures 25, 739–755.
- Gafsi M., Brossier J. (1997) Farm management and protection of natural resources: Analysis of adaptation process and the dependence relationships, Agr. Syst. 55, 71–97.
- Giampietro M. (1997) Linking technology, natural resources, and the socio-economic structure of human society: A theoretical model, Adv. Human Ecol. 6, 75–130.
- Gibon A., Sibbald A.R., Flamant J.C., Lhoste P., Revilla R., Rubino R., Sorensen J.T. (1999) Livestock farming systems research in Europe and its potential contribution for managing towards sustainability in livestock farming, Livest. Prod. Sci. 61, 121–137.
- Gotsch N., Rieder P. (1990) Forecasting future developments in crop protection, Crop Prot. 9, 83–89.
- Gowdy J. (2007) Avoiding self-organized extinction: Toward a co-evolutionary economics of sustainability, Int. J. Sust. Dev. World Ecol. 14, 27–36.
- Grumbine R.E. (1994) What is ecosystem management? Conserv. Biol. 8, 27–38.
- Gueringer A., Rapey H., Houdart M., Bigot G., Josien E., Landré F. (2009) Adaptability through spatial management. A case study of livestock farms in the Massif Central, France, Outlook Agr. 38, 111–118.
- Hagmann J., Chuma E. (2002) Enhancing the adaptive capacity of the resource users in natural resource management, Agr. Syst. 73, 23–39.
- Hart R.D., Pinchinat A.M. (1982) Integrative agricultural systems research, in: Servant J., Pinchinat A. (Eds.), Caribbean Seminar on Farming Systems Research Methodology, Pointe-à-Pitre, Guadeloupe, FWI, May 4–8, 1980, INRA-IICA Ed., pp. 555–565.
- Holland J.H. (2006) Studying complex adaptive systems, J. Syst. Sci. Complex. 19, 1–8.
- Holling C.S. (2001) Understanding the complexity of economic, ecological, and social systems, Ecosystems 4, 390–405.
- Holling C.S., Meffe G. (1996) Command and control and the pathology of natural resource management, Conserv. Biol. 10, 328–337.
- Holling C.S., Gunderson L.H., Ludwig D. (2002) In quest of a theory of adaptive change, in: Gunderson L.H., Holling C.S.

(Eds.), Panarchy: Understanding Transformations in Human and Natural Systems, Island Press, Washington DC, pp. 3–22.

- Hubert B., Ison R., Röling N. (2000) The 'problematique' with respect to industrialised-country agricultures, in: Cerf M., Gibbon D., Hubert B., Ison R., Jiggins J., Paine M., Proost J., Röling N. (Eds.), Cow up a tree, Learning for change in agriculture – Case studies from industrialised countries, Paris: INRA, pp. 13–29.
- IAASTD (2009) Agriculture at a crossroads, Synthesis report. International Assessment of Agricultural Knowledge, Science and Technology for Development, Island Press, Washington DC.
- Ison R.L., High C., Balckmore C.P., Cerf M. (2000) Theoretical frameworks for learning-based approaches to change in industrialised-country agricultures, in: Cerf M., Gibbon D., Hubert B., Ison R., Jiggins J., Paine M., Proost J., Röling N. (Eds.), Cow up a tree. Knowing and learning for change in agriculture, Case studies from industrialised countries, INRA, Paris, pp. 31–53.
- Ison R.L., Maiteny P.T., Carr S. (1997) Systems methodologies for sustainable natural resources research and development, Agr. Syst. 55, 257–272.
- Jacobson C., Hughey K.F., Allen W.J., Rixecker S., Carter R.W. (2009) Toward more reflexive use of adaptive management, Society and Natural Resources 22, 484–495.
- Jiggins J., Röling N. (1994) Systems thinking and participatory research and extension skills: Can these be taught in the classroom? Occasional Papers in Rural Extension, No. 10, Dept. of Rural Extension Studies, University of Guelph.
- Kahneman D. (2003) Maps of bounded rationality: Psychology for behavioural economics, Am. Econ. Rev. 93, 1449–1475.
- Lee K.N. (1999) Appraising adaptive management, Conservation Ecology 3, 3, [(online)] URL: http://www. ecologyandsociety.org/vol3/iss2/art3/.
- Lemery B., Ingrand S., Dedieu B., Degrange B. (2005) Agir en situation d'incertitude : Le cas des éleveurs bovins allaitants, Econ. Rurale 288, 57–69.
- Lev L., Campbell D. (1987) The temporal dimension in Farming Systems Research: The importance of maintaining flexibility under conditions of uncertainty, J. Rural Studies 3, 123–132.
- Levin S.A. (1998) Ecosystems and the biosphere as complex adaptive systems, Ecosystems 1, 431–436.
- Levrouw F., Morales H., Arbeletche P., Malaquin I., Tourrand J.F., Dedieu B. (2007) Les élevages uruguayens, le long terme et les incertitudes : une diversité de stratégies pour se maintenir dans la production, Rencontres Recherches Ruminants 14, 413–416.
- Long N. (1984) Creating space for change. A perspective on the sociology of development, Sociologia Ruralis 24, 168–184.
- Lopez-Ridaura S., van Keulen H., van Ittersum M., Leffelaar P.A. (2005) Multiscale methodological framework to derive criteria and indicators for sustainability evaluation of peasant natural resource management systems, Environ. Dev. Sustain. 7, 51–69.
- Macé K., Morlon P., Munier-Jolain N., Quéré L. (2007) Time scales as a factor in decision-making by French farmers on weed management in annual crops, Agr. Syst. 93, 115–142.
- Madelrieux S., Dedieu B. (2008) Qualification and assessment of work organisation in livestock farms, Animal 2, 435–447.

- Magne M.-A., Cerf M. (2009) How information becomes a resource for action in an uncertain and complex world, Outlook Agr. 38, 157–165.
- Manson S.M. (2001) Simplifying complexity: a review of complexity theory, Geoforum 32, 405–414.
- Mayumi K., Giampietro M. (2001) The epistemological challenge of modelling sustainability: Risk, uncertainty and ignorance, Paper prepared for FRONTIERS 1, held 4–7 July 2001 at New Hall, Cambridge, UK.
- Millennium Ecosystem Assessment (2005) Ecosystems and Human Well-being: Synthesis, Island Press, Washington, DC.
- Meyer A.D., Gaba V., Colwell K.A. (2005) Organizing far from equilibrium: Nonlinear changes in organizational fields, Organization Sci. 16, 456–473.
- Norman D.W. (2002) The farming systems approach: A historical perspective, Presentation held at the 17th Symposium of the International Farming Systems Association in Lake Buena Vista, Florida, USA, 17–20 Nov. 2002.
- Norman D.W., Malton P.J. (2000) Agricultural systems research and technical change, in: Colin J.-P., Crawford E.W. (Eds.), Research on Agricultural Systems. Accomplishments, Perspectives and Issues, Nova Science Publishers, pp. 17–47.
- Norton G.A. (1976) Analysis of decision making in crop protection, Agro-Ecosystems 3, 27–44.
- OECD (2009) The role of agriculture and farm household diversification in the rural economy: Evidence and initial policy implications, Document TAD/CA/APM/WP(2009)1/ FINAL, Organisation for Economic Co-operation and Development, Paris.
- Okey B.W. (1996) Systems approaches and properties, and agroecosystem health, J. Environ. Manage. 48, 187–199.
- Olsson P., Folke C. (2001) Local ecological knowledge and institutional dynamics for ecosystem management: A study of Lake Racken Watershed, Sweden, Ecosystems 4, 85–104.
- Ondersteijn C.J., Giesen G.W., Huirne R.B. (2006) Perceived environmental uncertainty in Dutch dairy farming: The effect of external farm context on strategic choice, Agr. Syst. 88, 205–226.
- Østergård H., Finckh M., Fontaine L., Goldringer I., Hoad S., Kristensen K., Lammerts van Bueren E., Mascher F., Munk L., Wolfe M. (2009) Time for a shift in crop production: Embracing complexity through diversity at all levels, J. Sci. Food Agr. 89, 1439–1445.
- Osty P.L. (1978) L'exploitation agricole vue comme un système, Bulletin Technique d'Information, Paris : Ministère de l'Agriculture.
- Park J., Seaton R.A. (1996) Integrative research and sustainable agriculture, Agr. Syst. 50, 81–100.
- Penrose E. (1997) The theory of the growth of the firm, in: Foss N. (Ed.), Resources, firms and strategies. A reader in the resource-based perspective, Oxford University Press, Oxford, pp. 27–39.
- Petit M. (1978) The farm household complex as an adaptive system, Proceedings of the 4 Forschungscolloquium des Lehrstuhls für Wirtschaftslehre des Landbaus, Arbeitsbericht 78/1, Institut für Landwirtschaftliche Betriebs- und Arbeitslehre, University of Kiel, pp. 57–70.

- Petit M. (1981) Théorie de la décision et comportement adaptatif des agriculteurs, Formation des agriculteurs et apprentissage de la decision, Dijon, ENSSAA, INPSA, INRA, INRAP.
- Puilliez L., Martin O., Tichit M., Sauvant D. (2008) Simple representation of physiological regulations in a model of lactating female: Application to the dairy goat, Animal 2, 235–246.
- Rammel C. (2003) Sustainable development and innovations: Lessons from the Red Queen, Int. J. Sust. Dev. 6, 395–416.
- Rammel C., van den Bergh J. (2003) Evolutionary policies for sustainable development: adaptive flexibility and risk minimising, Ecol. Econ. 47, 121–133.
- Rammel C., Stagl S., Wilfing H. (2007) Managing complex adaptive systems – A co-evolutionary perspective on natural resource management, Ecol. Econ. 63, 9–21.
- Rathe K., Witt U. (2001) The nature of the firm Static versus developmental interpretations, J. Manage. Governance 5, 331–351.
- Reix R. (1979) La flexibilité de l'entreprise, Éditions Cujas, Paris.
- Sánchez E., Sifres A., Casañas F., Nuez F. (2008) The endangered future of organoleptically prestigious European landraces: Ganxet bean (*Phaseolus vulgaris* L.) as an example of a crop originating in the Americas, Genet. Resour. Crop Ev. 55, 45–52.
- Scoones I., Thompson J. (1994) Beyond farmer first: rural people's knowledge, agricultural research and extension practice, London: Intermediate Technology Publications.
- Seath G.W., Webby R.W. (2000) The results and success factors of a farm monitoring and study group approach to collective learning in New Zealand, in: Cerf M., Gibbon D., Hubert B., Ison R., Jiggins J., Paine M., Proost J., Röling N. (Eds.), Cow up a tree. Knowing and learning for change in agriculture, Case studies from industrialised countries, INRA, Paris, pp. 111–119.
- Simon H.A. (1986) Rationality in Psychology and Economics, J. Business 59, S209–S224.
- Slaughter D.C., Giles D.K., Downey D. (2008) Autonomous robotic weed control systems: A review, Comput. Electron. Agr. 61, 63–78.
- Smit B., Wandel J. (2006) Adaptation, adaptive capacity and vulnerability, Global Environ. Chang. 16, 282–292.
- Smit H.T., Trigeorgis L. (2006) Strategic planning: Valuing and managing portfolios of real options, R&D Management 36, 403–419.
- Stewart P. (2001) Complexity theories, social theory and the question of social complexity, Philos. Soc. Sci. 31, 323–360.
- Sumberg J., Okali C., Reece D. (2003) Agricultural research in the face of diversity, local knowledge and the participation imperative: Theoretical considerations, Agr. Syst. 76, 739–753.
- Tarondeau J.-C. (1999) Approches et formes de flexibilité, Revue Française de Gestion 123, 66–71.
- Teece D.J. (2007) Explicating dynamic capabilities: The nature and microfoundations of (sutainable) enterprise performance, Strategic Manage. J. 28, 1319–1350.
- Tichit M., Ingrand S., Moulin C.H., Cournut S., Lasseur J., Dedieu B. (2004) Analyser la diversité des trajectoires productives des femelles reproductrices : intérêts pour modéliser le fonctionnement du troupeau en élevage allaitant, INRA Productions Animales 17, 123–132.

- Tilman D., Cassman K.G., Matson P.A, Naylor R., Polasky S. (2002) Agricultural sustainability and intensive production practices, Nature 418, 671–677.
- Tsai Y.J., Mishoe J.W., Jones J.W. (1987) Optimizing multiple cropping systems: Simulation studies, Agr. Syst. 25, 165–176.
- Trigeorgis L. (2005) Making use of real options simple: An overview and applications in flexible/modular decision making, Eng. Econom. 50, 25–53.
- Urry J. (2005) The complexities of the global, Theory, Culture and Society 22, 235–254.
- van den Bergh J.C., Ferrer-i-Carbonell A., Munda G. (2000) Alternative models of individual behaviour and implications for environmental policy, Ecol. Econ. 32, 43–61.
- van den Bergh J.C. (2007) Evolutionary thinking in environmental economics, J. Evolution. Econ. 17, 521–549.
- Walker B., Gunderson L., Kinzig A., Folke C., Carpenter S., Schultz L. (2006) A handful of heuristics and some propositions for understanding

resilience in social-ecological systems, Ecology and Society 11, 13, [(online)] http://www.ecologyandsociety.org/vol11/ iss1/art13/.

- Walsh J.P., Meyer A.D., Schoonhoven C. (2006) A future for organization theory: Living in and living with changing organizations, Organization Sci. 17, 657–671.
- Westley F. (2002) The devil in the dynamics: Adaptive management on the front lines, in: Gunderson L.H., Holling C.S. (Eds.), Panarchy, Understanding Transformations in Human and Natural Systems, Island Press, Washington DC, pp. 333–360.
- Wilson G.A. (2008) From 'weak' to 'strong' multifunctionality: Conceptualising farm-level multifunctional transitional pathways, J. Rural Studies 24, 367–383.
- Wolfe M.S., Baresel J.P., Desclaux D., Goldringer I., Hoad S., Kovacs G., Löschenberger F., Miedaner T., Østergård H., Lammerts van Bueren E.T. (2008) Developments in breeding cereals for organic agriculture, Euphytica 163, 323–346.

# **Economics of Biosecurity Across Levels of Decision-Making**

Jaakko Heikkilä

## Contents

1	Introduction					
2	Biosecurity					
3	Economics of Biosecurity					
	3.1	Reviews	63			
	3.2	Category 1: Prioritisation Between Different				
		Types of Hazards	65			
	3.3	Category 2: Prioritisation Between Different				
		Types of Biological Hazards	65			
	3.4	Management Strategies, Methods and Policies .	66			
	3.5	Category 4: Cost Estimates of Disease				
		Outbreaks and Species Invasions	73			
4	Disc	ussion	76			
References						

J. Heikkilä (🖂)

MTT Economic Research, Latokartanonkaari 9, 00790 Helsinki, Finland e-mail: jaakko.heikkila@mtt.fi

Abstract Biosecurity is a concept that has important economic, social, ecological and health-related dimensions. By biosecurity we mean protection of production, ecosystems, health and the social infrastructure from external threats caused by pests, pathogens and diseases of various forms and origins. The fact that more goods, transport platforms and people are moving around the globe at increasing speeds provides unforeseen possibilities for rapid spread of different types of organisms. This is exacerbated by changes in the production structures and climate. As a result, both the benefits and the risks of changes in the food system cross borders more often, leading to an increased demand for biosecurity policies. Economics can be related to biosecurity in at least three fundamental ways. First, many of the ultimate or proximate causes of bioinvasions create economic welfare. Second, bioinvasions result in various types of impacts, many of which are economic by nature - or at least may be measured in economic terms. Third, the negative impacts of invasions or their probability of occurrence can often be either avoided or reduced. These biosecurity policies themselves have economic implications, which often may be quite different from those caused by the biological hazard itself. A few reviews of separate components of economics of biosecurity exist, but there have been no attempts to review the big picture. Instead, the previous reviews have concentrated on different components of biosecurity such as invasive species or animal diseases. Our aim is to look at the issue in broad terms, draw some commonalities from the research conducted, and identify areas in which economic analyses have primarily been conducted and in which areas there remains work to do. The review includes about 230 studies from all areas of biosecurity

up to the year 2008. The review finds that study of economics of biosecurity is growing steadily, but is still relatively concentrated on narrow questions, few countries, few species/diseases and few journals.

**Keywords** Biosecurity • Economics • Invasive Species • Pests • Diseases • Review

## **1** Introduction

Biosecurity is a concept that has important economic, social, ecological and health-related dimensions. By biosecurity we mean protection of production, ecosystems, health and the social infrastructure from external threats caused by pests, pathogens and diseases of various forms and origins. In this paper, we call these external biological threats collectively 'biological hazards'. An entry of a biological hazard is referred to as an 'invasion'.

The ultimate global forces that accelerate invasions by biological hazards include changes in production structure and land use, climatic change and globalisation (see, e.g., Daily and Ehrlich, 1996). In this context globalisation refers to the increasing integration of markets for goods, services and capital throughout the world. This leads to an increased volume of transactions and reduced trade barriers and transaction costs, all driven by the gains of trade. The fact that more goods, transport platforms and people are moving around the globe at increasing speeds also provides unforeseen possibilities for rapid spread of different types of organisms. This is exacerbated by changes in the production structures and climate. As a result, both the benefits and the risks of changes in the food system cross borders more often, leading to an increased demand for biosecurity policies.

Some of the organisms that are moved provide positive economic impacts on the receiving economic system, and some provide negative economic impacts. Some organisms produce both (often intended) positive impacts as well as (often unintended) negative impacts, with the net impact being sometimes difficult to determine. Nonetheless, the fact that species transfer is an issue that cannot be defined as dominantly beneficial or dominantly harmful makes biosecurity policy challenging to design and implement.

Economics is related to biosecurity in at least three fundamental ways. First, many of the ultimate or proximate causes of bioinvasions create economic welfare. Be it related to transport, tourism and trade, or to urbanisation, migration and agricultural intensification, many outcomes of economic decision-making provide incentives or means for species and diseases to be moved around the world. Second, bioinvasions result in various types of impacts, many of which are economic by nature - or at least may be measured in economic terms. These include damage to production processes, health and the environment (see, e.g., FAO, 2001). Third, the negative impacts of invasions or their probability of occurrence can often be either avoided or reduced. These biosecurity policies themselves have economic implications, which often may be quite different from those caused by the biological hazard itself.

A few reviews of separate components of economics of biosecurity exist, but there have been no attempts to review the big picture. Instead, the previous reviews have concentrated on different components of biosecurity such as invasive species (Born et al., 2005; Gren, 2008; Heikkilä, 2006; Olson, 2006) or animal diseases (Elbakidze, 2003). Our aim is to look at the issue in broad terms, draw some commonalities from the research conducted, and identify areas in which economic analyses have primarily been conducted and in which areas there remains work to do.

The review includes 231 studies from all areas of biosecurity up to the year 2008. The review finds that study of economics of biosecurity is growing steadily, but is still relatively concentrated on narrow questions, few countries, few species/diseases and few journals. The following priority themes (in no particular order) are suggested for further study: (1) sophisticated economic policy instruments to mitigate biological hazards; (2) rigorous economic prioritisation frameworks; (3) clearer understanding of the distribution of biosecurity impacts and the associated incentives; (4) closer links with the costs and methods of prevention to the economic risk; and (5) the role of human behaviour in biosecurity and how that behaviour can best be directed.

The rest of the paper is organised as follows. In Section 2 we briefly define biosecurity and economics of biosecurity in order to clarify which questions and topics are included in the review. In Section 3 we first discuss existing reviews, and then review the existing studies based on the framing of the research question. In Section 4 we draw some conclusions as well as give recommendations for further study.

## 2 Biosecurity

Biosecurity provides an appropriate framework for managing the risks presented by biological hazards. It can be defined as the exclusion, eradication and effective management of risks posed by pests and diseases to the economy, environment and human health (Biosecurity Council, 2003). The main targets for biosecurity policies include environmental, commercial, cultural, human health-related, and social objectives. The FAO (2003) sees biosecurity as the "process and objective of managing biological risks associated with food and agriculture in a holistic manner". Policy is here defined as any instrument or institution that is working to reduce the risks (i.e. either probability or impact) of biological hazards. Although parts of biosecurity, this review excludes studies on genetically modified organisms (GMOs) and diseases that are confined solely to humans.

The four main elements that biosecurity policies protect include human health, animal health, plant health and environmental health, including the health of the built environment. Each category will be shortly discussed below. However, it is worth noting that many biological hazards are not limited to a certain category. Instead, a particular species or disease may cause detrimental impacts in several ways and in several sectors. For instance, avian influenza is of concern for human, animal and environmental health.

The human health aspect of biosecurity includes zoonotic diseases (diseases communicable between animals and humans) and biological food safety. Hazards may be divided into bacteria (e.g. salmonella), viruses (e.g. avian influenza and HIV/AIDS), protozoa (e.g. the malaria parasite *Plasmodium* spp.), fungi and worms (e.g. trichinosis), and prions (e.g. bovine spongiform encephalopathy, BSE) (Delfino and Simmons, 2000). The importance of biological hazards that threaten human health is vast (see e.g. Wolfe et al., 2007). For instance, smallpox was perhaps the largest single cause of death in the 20th century, and the Spanish flu of 1918 (H1N1-type avian influenza) killed approximately 40–50 million people and attacked the young disproportionately (IMF, 2006). Today, on a global scale communicable diseases are estimated to cause about a quarter of all mortality and morbidity, and even in industrial countries one-third of the population acquire annually a disease caused by a food-borne microbe (WHO, 2007).

The human health component includes not only physical/biological health, but also social health. This refers to human wellbeing separate from mere infection by a disease. For instance, many animal diseases and government policies related to animal diseases are likely to affect social health. This may be due to, for example, movement restrictions that result in social isolation, or disposal of infected animal carcasses (e.g. Rossides, 2002; Mort et al., 2005; Dixon, 2007). Welfare reduction may also result from loss of cultural heritage and spiritual or religious values when, for instance, a traditional ecosystem is transformed into an ecosystem dominated by exotic species.

Animal health is another component of biosecurity, including animal diseases as well as animal predators, parasites and parasitoids. The importance of animal health is likewise vast. For instance, in North America and Europe calories of the human diet are mainly obtained from animal products (Carvalho, 2006). Examples of diseases threatening animal health include classical swine fever, rinderpest, Newcastle diseases and Aujeszky's disease. In addition, zoonotic diseases are an animal health issue as well as a human health issue. For instance, Ebola haemorrhagic fever has caused the death of thousands of gorillas in Gabon and the Republic of Congo (Bermejo et al., 2006).

The plant health aspect of biosecurity includes plant diseases and pests. The injurious effect may be direct, as in the case of pests that directly affect the plant, such as pathogens, parasites and herbivores, but it may also be indirect, as in the case of weeds that compete for biotic or abiotic resources, or any other organism impacting on the plant indirectly (Schrader and Unger, 2003). Plant production provides the basis of the human food chain, and it is therefore a crucial target of biosecurity. In addition, forest products are a source of significant revenue for many countries, both in the developed world (timber, recreation, biodiversity, berries) as well as in the developing world (forage, construction, fuelwood, biodiversity, medicine). About 75% of calories in a balanced human diet are from plant products (Carvalho, 2006). For instance, the famine in Ireland in the 1840s was caused by *Phytophtora infestans* – a

fungus that arrived from North America and destroyed the Irish potato harvest (Schrader and Unger, 2003).

Health of the environment and built environment is the fourth component of biosecurity. Invasive alien species may be harmful to environmental health through various kinds of biotic or abiotic interactions (including, for instance, predation, competition, interbreeding and spread of diseases). For example, the Arundo donax giant reed, which is currently under consideration for bioenergy production in Europe, has been found to have detrimental ecosystem impacts in California riversides (Hundley, 2007). We also include the built environment within environmental health. Examples are provided by, for instance, damage to roads, runways and parks caused by rabbits (Gebhardt, 1996) or damage to power-plant intake pipes by the zebra mussel (Pimentel et al., 1999, 2005; Connelly et al., 2007).

The aspects of biosecurity discussed above, as well as the concepts of biosecurity, food safety and dietary safety are clarified in Fig. 1. The figure illustrates that there are components of food safety not included within the definition of biosecurity, as well as several other factors affecting the health of humans. For instance, human health is partly affected by biosecurity (e.g. foodborne microbes and zoonotic diseases), partly by non-biological aspects of food safety (e.g. acute chemical substances), partly by dietary safety (e.g. accumulating chemical substances), partly by the way of life not related to the environment or biological hazards (e.g. sports, stress), and partly by the environmental factors surrounding us (e.g. noise, temperature). While it is true that, for instance, exotic plant pests can drive changes in risks of natural forces (e.g. fire regime), such feedbacks are not presented in the figure.

Biosecurity is primarily about managing risk and uncertainty. Risk in economic terms is a continuum of possible events and future states of the world. Put simply, the economic risk is the probability of an event (or future path) multiplied by the economic costs and benefits that ensue if that path materialises. The risk is thus affected by the likelihood of an event as well as the net cost (costs less benefits) of that event. The probability and the impacts are usually not known with certainty.



Fig. 1 Biosecurity, food safety and dietary safety

Uncertainty may be divided into uncertainty regarding biological and natural processes and uncertainty regarding human factors (Heal and Kriström, 2002). Regarding natural processes, there may be uncertainty and natural stochastic variation in the invasion process and subsequent spread processes. Uncertainty regarding the human factors arises from human preferences and decision-making as well as from the functioning of the society and its institutions. These include issues such as how biosecurity hazards and policies affect human systems, and how some unknown economic values (such as the value of life or the discount rate) affect the evaluation. Thus, even if we knew the science behind certain natural science phenomena with certainty, the impacts of such phenomena on the various components of human societies may remain uncertain. For instance, it is unknown how the continued presence of the biosecurity hazards would affect the production patterns or land use in the longer run. In addition, there may be uncertainty related to which policies are needed to address the problems, how those policies impact on the issue in question and what the costs of undertaking these policies are (Heal and Kriström, 2002).

An important implication here is that the risk is an endogenous process that is affected not only by the biological processes, but also by the human activities that create the risk as well as human reactions to the risk. For instance, it has been shown that socio-economic variables are significant in explaining the invasion process in the context of plants (Dehnen-Schmutz et al., 2007; Dalmazzone, 2000), marine species (Weigle et al., 2005), insects (Work et al., 2005) and infected meat imports (Wooldridge et al., 2006). So risk is as much an economic question as a biological one (Mumford et al., 2000). It is a matter of allocating resources so that the negative effects of risk are reduced enough compared with efforts required to reduce the risk.

Despite risk and uncertainty, we would like to maximise social net benefits. There are several actions that we may take to deal with risk. First, risk may be avoided. This can be achieved by, for instance, not using (or trading) a potentially risky good and thus foregoing the utility from the good. Second, the realisation of the risk may be avoided. This requires increased surveillance at all levels, extra care in the use of the good, and quality management in the entire production and consumption chain. Third, we may accept the presence of the risk. This means getting prepared to face the consequences and manage them in the case that the risk materialises. Fourth, risk may be transferred by, for instance, hedging, insuring or diversifying.

Biosecurity thus involves protection of various environmental and anthropogenic elements in an uncertain world against the threat presented by biological hazards. The primary question in most economic studies (albeit not explicitly stated in most of them) is: what should we do about this risk?

#### **3 Economics of Biosecurity**

## 3.1 Reviews

There have been a few reviews of separate sections within the economics of biosecurity. Olson (2006) reviews economics of terrestrial invasive species. He suggests that there has been rapid development in the field, but uncertainty, spatial modelling, prevention, trade, and conflict between private and public incentives are areas where more sophisticated analyses are needed. Gren (2008) reviews invasive species studies in relation to two questions: (1) how to set targets for species damage mitigation; and (2) which policy instruments are best in achieving the targets. Her results indicate that strategies for prevention, control and damage reduction are complementary, and ignoring any one of them may impose social costs. She also argues that there is a lack of empirical applications and of studies evaluating current regulations.

There are also two reviews on cost-benefit types of studies. Born et al. (2005) review the economic cost-benefit evaluations of invasive species. Their main conclusions are that: (1) studies mostly have methodological shortcomings; (2) assessments are mostly expost rather than ex-ante; (3) prevention is hardly reflected in the analyses; and (4) uncertainty is insufficiently addressed. They conclude that most studies "focus on ex-post evaluation, on control measures, on few countries, on agriculture, and on use values" (Born et al., 2005).

Similar conclusions are reached in a review by Heikkilä (2006), who makes three basic observations on studies on invasive species in an agricultural context. First, the policy alternatives evaluated are often simply undertaking some policy versus not undertaking it. The most common analysis seems to be between a current protective policy versus abandoning it. Wider policy options are generally either not available or not analysed. Second, it is often only the direct, easily monetised costs that are included in the quantitative analysis. Costs that are more difficult to analyse - such as trade or employment effects - are often ignored altogether, even though in the cases where they are included they turn out to be very important. Third, sensitivity analyses and treatment of uncertainty are in many cases inadequate, if conducted at all. Heikkilä (2006) also notes that in many cases invasive species cost-benefit studies are not written by economists. The same can be argued to apply to research on animal diseases. In fact, Adamson and Cook (2007) go so far as to argue that standardised approaches to economic evaluations have not been developed and economic reviews are generally undertaken by consultants. As a result, the economic foundation of the studies is not as strong as it could be.

Elbakidze (2003) reviews the economics of agricultural biosecurity in relation to bioterrorism and argues that especially in relation to foot and mouth disease surveillance and detection (early actions) are overlooked in studies compared to vaccination and slaughter (reactive actions) – an idea shared by Born et al. (2005) in relation to invasive species. McInerney et al. (1992) concluded that economic studies on livestock disease concentrate on cost-benefit analyses of control programmes or gross estimates of costs of particular diseases, neither of which offers an adequate basis for economic decisions. There are also reviews on economics of specific biological hazards; for instance, on bovine viral diarrhoea virus (Houe, 2003) and parasitic diseases of production animals (Perry and Randolph, 1999).

Economics is about the scarcity of resources, and how to allocate the available resources in the best possible way to maximise the objectives set by the decision-maker. It has been noted that the role of economics can be to evaluate a chosen policy, to identify the most cost-effective measures to attain a specified goal, or to prioritise between different goals and measures (Goldbach and Alban, 2006). Embedded in this notion is the fact that there are various hierarchical levels at which economic decisions need to be made. Ideally, research that attempts to answer such questions would involve simultaneous considerations of the management strategy, objectives within the strategy and the tools to achieve those objectives in the best possible manner. The objective, after all, should be to maximise (social) net benefits subject to given constraints.

However, individual papers rarely discuss the whole management issue. Instead, the research question is often limited to, for instance, evaluation of specific management alternatives. Therefore, in this review the existing studies are organised categorically, based on the aims of the research setting in which they study biosecurity economics. The first category is priority-setting between different kinds of hazards confronting human society including, for instance, biological, chemical, social and military hazards. The second category is priority-setting between different kinds of biological hazards; for instance, between invasive species and animal diseases, which are often administered in separate ministries. The third category is divided into three subcategories. The first subcategory is evaluation of management strategies to deal with a biological hazard; for instance, prevention versus adaptation. The second subcategory is the costefficiency of different management methods; for instance, questions such as the optimal tax level or the optimal pre-emptive slaughter radius around infected premises. The third subcategory deals with the evaluation of the available policy instruments aiming to achieve the desired outcome of the chosen management strategy - for instance, whether we should use economic instruments or command and control instruments. Finally, the fourth category deals with the assessment of impacts (usually costs) of specific invasions or outbreaks, and is often done given the prevailing strategies, policy instruments and management methods. Such assessments usually do not include any comparisons but rather an outright monetary evaluation of the incurred or simulated costs.

These categories are presented in Fig. 2 and reviewed in more detail below. Additionally, two examples are discussed. The first example deals with a very common setting in management strategy assessment of category 3, namely the choice between prevention versus adaptation strategy. The second example deals with import risk, and discusses the types of studies dealing with this risk. Whilst ideally the question would be tackled as an entity, the examples demonstrate how individual studies often address more specific questions.



Fig. 2 Categorisation of economic biosecurity studies

# 3.2 Category 1: Prioritisation Between Different Types of Hazards

Renn (2006) divides hazardous agents into biological agents, physical agents (radiation, temperature, noise), chemical agents (pollutants, carcinogens), natural forces (drought, flood, tsunami, hurricane, fire, avalanche, earthquake), socially communicative hazards (terrorism, sabotage, mass hysteria, violence), and complex hazards (combinations of the above). How the society's resources are divided among these hazards is to a large extent a political question, and has not to our knowledge been studied academically. Having said that, we do not see any intrinsic reasons why such an exercise could not be undertaken. Naturally, comparison and estimation of risks and costs associated with such a wide variety of hazards is complex, and no easy solutions for comparison are available. The risks and impacts of such hazards may show such variety that they are, in addition to being difficult to value, very difficult to compare with one another. For instance, comparing the impacts of a materialising military hazard with the loss of welfare from burning animal carcasses may prove to be a daunting exercise.

Yet, resource allocations are constantly made at the level of the government as decisions regarding allocation of resources between, say, ministries of the environment (invasive species) and agriculture (health of production animals and plants) are made. Studies on individual hazards can achieve deeper understanding of the specific circumstances, but it would be worthwhile having analyses of the larger issues and tradeoffs present at higher levels to assist such resource allocation issues. For instance, how resources should be allocated regarding chemical versus biological hazards in food products would already be a prioritisation question in this category. This has, indeed, been studied to some extent (Valeeva et al., 2007, 2006).

# 3.3 Category 2: Prioritisation Between Different Types of Biological Hazards

In this category the focus is purely on biological hazards. These can be divided into hazard categories as discussed in Section 2 (human, animal, plant and environmental health). Economic prioritisation can take place at two levels: (1) between different categories and (2) within the categories. As far as we are aware, prioritisation between the categories – for instance, comparing the economic risk presented by animal diseases to risks posed by plant diseases – has not taken place. However, for comparison of benefits from protection against different kinds of biological hazards, see Waage et al. (2005).

Instead, some comparisons within the hazard categories have taken place. In fact, such actions are also already politically sought for within animal diseases: the European Union is seeking to harmonise animal health policy and is looking for a single legal framework to bring together trade, imports, animal disease control, nutrition and welfare. It aims to introduce a scheme for classifying and prioritising animal diseases and disease prevention according to their economic or human health impact (European Commission, 2006). Somewhat similar plans are underway regarding plant health and invasive alien species and, for instance, risk-based surveillance overall is becoming increasingly popular (McKenzie et al., 2007), but these trends have not yet materialised as clearly as in the case of animal health.

The problem in lack of such assessments is that the resource allocation may be inefficient. As some kind of allocation is in any case made, if it is not based on (economic) assessment of the risks, it is based on something else. For instance, Virtue (2007) points out that in Australia too many species have been declared weeds in order to effectively contain or eradicate them all. Australian regions are argued to focus on widespread species for various reasons, including a long history of control, high visibility, political pressure, perceived impacts, shared burden of control, familiarity, and agricultural bias. Economic prioritisation has rarely played a part in such choices (Virtue, 2007). Similarly, risk is not the primary determining factor in control of many animal diseases (Rosengren and Heikkilä, 2009).

A few rigorous economic studies exist that prioritise hazards within certain categories. In addition, there are several studies that include some sort of risk ranking of biological hazards, including their economic impacts, but where the methodologies are less rigorous. In the first of the two more rigorous studies that we are aware of, Moffitt and Osteen (2006) study prioritisation of invasive species threats under uncertainty. They examine different decision rules and develop a priority list of potential agricultural crop pests in the United States to demonstrate the criteria. The other example is Cook (2005), who studies the critical level of investment in pest exclusion in Western Australia. He provides a framework that allows prioritisation of pests based on damage and production cost increases, and also identifies a critical level of expected damage associated with the pest that can be used as a ceiling for incursion expenditure. If this level is exceeded, other control options such as containment or adaptation should be considered. Using a static partial equilibrium model, the study considers a range of host plants.

Various hazard ranking studies are usually somewhat less rigorous from an economic perspective. This type of ranking usually follows the basic structure of Covello-Merkhofer risk assessment, including release, exposure and consequence assessments, which are combined to form a risk estimate (Peeler et al., 2007). The approach has been applied to, for instance, weeds in the US (Parker et al., 2007) and in Europe (Weber and Gut, 2004), to ants (Harris, 2005) and wildlife pathogens (McKenzie et al., 2007) in New Zealand, and to plant pests in Western Australia (Cook and Proctor, 2007). In these assessments the basic themes relate to what components are included in the risk ranking, how they are scored and aggregated, and how the opinions are formed.

Hazard ranking is based on separate components that together form the ranking order. The components typically include the probability of entry (or invasion or introduction or outbreak), the probability of spread (or invasiveness), and the likely consequences, which may or may not be measured in monetary terms. In most cases the ranking is on an arbitrary scale (e.g. 1 to 5), and the overall rank is obtained by summing up the different component values (e.g. Harris, 2005). Also, multiplicatory scoring is used, in which case the rank approaches zero if any of the individual components does so. This approach may be augmented by a measure for the feasibility of control. For instance, Virtue (2007) augments the risk measure by a containment feasibility measure, which is the product of control costs, current distribution and persistence. Rank is then obtained by dividing the risk measure by the control feasibility measure. The priority assessment is usually done using either expert opinion on its own (Weber and Gut, 2004; Cook and Proctor, 2007) or combined with a literature review (McKenzie et al., 2007). As pointed out by Cook and Proctor (2007), different juries are likely to produce different results.

Such prioritisation assessments are not without both theoretical and methodological problems, and the resulting rank order does not act as a decision-rule in itself. The rank order often illustrates the risk presented by the organism, not the benefit that would be achieved when investing a unit of control in it. Even so, a formal framework that approaches the problem in a standardised manner and helps in thinking through the issues involved is better than basing the decisions on, for instance, history, personal opinion or political pressure. A more detailed discussion on risk ranking is presented in Heikkilä (forthcoming).

# 3.4 Management Strategies, Methods and Policies

Study of biosecurity economics in category 3 is ideally a continuum and a holistic exercise in which the management strategy, the management methods to undertake that strategy and the policy instruments to obtain the desired outcome are evaluated simultaneously. However, this ideal is seldom achieved, at least in individual research papers. Instead, the choice between strategies, management methods or policy instruments is often made given the other factors. Hence in this review the studies are divided according to the objectives of the studies.

#### 3.4.1 Category 3a: Assessment of Management Strategies Within Hazards

Biological hazards may be managed through various strategies, including prevention, eradication, containment and control (CBD, 1992; Council of Europe, 2003; Shine et al., 2000). If prevention fails, the most extensive measure is to eradicate the entire invasive population or the disease. Within the realm of certain animal diseases and in the case of some plant pests or diseases, eradication may be an automatic action if the hazard is encountered. In the case of invasive species threatening the environment, such decisions are likely to be made much more on an ad hoc basis. For examples of eradication successes and failures, see Genovesi (2005) and Simberloff (2003). If eradication is found not to be feasible or economical, it is possible to try and contain the hazard to a given area and thus prevent any further spread. In containment, the objective is to protect the nearby areas, either inside the country in question or in neighbouring countries (Council of Europe, 2003). If this is also not possible, the final alternative is to control the hazard such that its population size, density or prevalence remains below some threshold level.

Although, for instance, the guiding principles of the United Nations Convention on Biological Diversity (CBD) advocate eradication as the next best thing if prevention fails, in the context of invasive species very few eradication programmes have been carried out in Europe. The reasons for this include inadequate national laws, unclear responsibilities among authorities and opposition by animal rights groups (Genovesi, 2005). In contrast, for animal diseases the case is somewhat different. Heath (2006) points out that all highincome countries have eradicated all new animal diseases provided that there has been political will to do so. For the poorer countries this is not the case, as private producers are not necessarily interested in diseases that cause large trade disruptions but where production losses may be modest. In other words, there is a divergence between private and social incentives.

In addition, invasions that are not targeted immediately when a hazard is first encountered and thus become widespread can be extremely costly to eradicate (see e.g. Cowan, 1992; Genovesi, 2005). Hence eradication is often feasible only when the size of the invading population is relatively small and it is geographically restricted (Council of Europe, 2003). Invasion dynamics are not the only issue to account for, however. Perrings (2000) establishes the conditions under which allowing establishment and spread of invasive species is optimal and emphasises that the relative costs and benefits of native and non-native species are important.

The society should explicitly account for the tradeoff between present expenditures to protect ourselves from the invasion, and future expenditures to control or reduce the damage from the few harmful invasions (Jensen, 2002). For instance, most studies on foot and mouth disease are argued to concentrate on post-outbreak disease management, and less attention has been paid to pre-event surveillance and detection. No studies have been conducted that assess the benefits of surveillance and detection relative to associated damage (Elbakidze and McCarl, 2005, see also Born et al., 2005).

Several economic studies exist where one management strategy (often the current strategy) is compared to an alternative strategy. Wider comparisons of various alternative strategies are much less common. Examples of comparisons for more than two strategies include, for instance, McInerney and Kooij (1997), and Rich and Winter-Nelson (2007). The most common strategy options that are evaluated include some sort of preventative action versus reactive control. These strategies have been studied by, for instance, Burnett et al. (2007), Fernandez (2008), Persson and Jendteg (1992), Maijala et al. (2005), Cembali et al. (2003), Ranjan et al. (2008), Cook et al. (2007), and Heikkilä and Peltola (2007).

Other economic strategy comparisons that have been undertaken include, for instance, eradication versus control (Eiswerth and van Kooten, 2002; Olson and Roy, 2002; Andersson et al., 1997), eradication versus containment (Cacho et al., 2008; Sharov and Liebhold, 1998; Sharov, 2004), detection versus control (Mehta et al., 2007), control versus no control (Fasina et al., 2007), vaccination versus alternatives (Mukhebi et al., 1999; Kobayashi et al., 2007; Berentsen et al., 1992), regionalisation versus no regionalisation (Paarlberg et al., 2007), and eradication with restoration versus adaptation (Zavaleta, 2000).

In most cases the strategy alternatives are seen as substitutes that are compared against each other. Much more rarely has a study incorporated strategy choices as complements in order to determine the optimal mix of strategies. Such a study was undertaken for prevention and control by Burnett et al. (2008) to determine the optimal intertemporal allocation of resources. Also, Pifafi and Rousmasset (2007) provide an integrated framework where optimal prevention is combined with optimal pest removal. They show how optimal prevention depends on minimised costs resulting from the failure of prevention, including damage costs. Other studies on complementary strategies include Burnett et al. (2006), Leung et al. (2002), and Kim et al. (2005).

The methods employed in these assessments show a wide (but thin) variety including, for instance, optimal control theory (Burnett et al., 2007), Monte Carlo simulation (Heikkilä and Peltola, 2007; Russell et al., 2006; Niemi et al., 2008), linear programming (Hastings et al., 2006) and game theory (Fernandez, 2008). Objects of the studies include a wide variety of different biological hazards.

The optimal management strategies have been found to vary in space and time. Location has been found to matter for the largest cost components (Persson and Jendteg, 1992) as well as for optimal strategies (Burnett et al., 2007). For instance, James and Rushton (2002) argue that in many studies (Netherlands, Italy, Germany, etc.) it has been shown that eradication of foot and mouth disease is more economical than vaccination in Europe, whereas in developing countries stamping out seems not to be feasible or desirable. Similarly, it is argued that in Nigeria, even at the expense of loss of export markets, the best method for H5N1 control is vaccination (Fasina et al., 2007). The optimal strategy has also been found to be time-varying (Hastings et al., 2006) as well as to depend on species characteristics (damage, growth rate) and varying from do nothing to a high level of effort (Mehta et al., 2007).

Costs that should be included in a thorough economic assessment of biosecurity management strategies, and that have variedly been included in the strategy assessments, comprise (after Burnett et al., 2007; Persson and Jendteg, 1992; Niemi et al., 2008; Cembali et al., 2003; Mehta et al., 2007; Kobayashi et al., 2007; Fasina et al., 2007): (1) infrastructure and administration cost; (2) search and detection cost; (3) cost of pest or disease control; (4) illness treatment costs (human, animal); (5) illness costs (human); (6) loss of productive capacity (human, animal, plant, environment); (7) cost of business interruption and demand switches; (8) price movements; and (9) compensation payments and other income transfers.

When the distribution of costs and benefits of management strategies has been studied, it has generally been found to be very uneven. For instance, Mukhebi et al. (1999) find that 91% of the costs of heartwater disease are borne by the commercial farmers and Cembali et al. (2003) find that the nurseries who pay for the analysed virus prevention programme receive benefits of \$0.5 million (0.2% of total benefits), whereas growers and consumers who do not pay for the programme benefit \$80 million and \$147 million, respectively. Similar findings are made by Gutrich et al. (2007) in the case of fire ant in Hawaii as well as Mangen and Burrell (2003) in the case of classical swine fever in the Netherlands. Also, Niemi and Pietola (2004) note that in the case of pig diseases, the privately optimal solution is not necessarily the one that is socially desirable.

Uncertainty is more and more included in the assessments in this category, but treatments of it as well as the implications of uncertainty differ widely. Uncertainty is important to account for, because it is often related to the outcome of the events, although not in a straightforward manner. It has, for instance, been found that uncertainty in initial population size (Mehta et al., 2007) and in timing and nature of invasion (Ranjan et al., 2008) has a substantial impact on the optimal strategy. On the other hand, Russell et al. (2006) find that in 60% of the cases considered in their study, the best policy did not change when uncertainty was introduced. Similarly, Jensen (2002) finds that his conclusions hold regardless of whether the costs are known for certain or only in distribution.

#### Example of Category 3a Problem Framing: Prevention Versus Adaptation

Prevention (or mitigation, avoidance, pre-emptive control) and adaptation (or reactive control, treatment, amelioration) are perhaps the most widely studied strategy alternatives. There is a general agreement that on most occasions preventative actions are the best strategy, given the uncertainties involved and the difficulties in eradicating most species and diseases reactively. For instance, Fernandez (2008) finds that cooperative and preventative abatement is optimal to all other strategies. This approach is put forward by, for instance, the intergovernmental scientific advisory body established by the CBD (Perrault and Carroll Muffett, 2001) as well as the European Union in relation to many animal diseases.

For instance, Leung et al. (2002) use stochastic dynamic programming to study how to devote resources between prevention and control efforts given uncertain invasion events. They apply their general model to the case of zebra mussel (*Dreissena polymorpha*) – an aquatic species that spread from Europe in ship ballast and has subsequently spread rapidly within the United States. The species damages power plants by blocking their water intake pipes. The authors show that the society could be made better off by spending up to 0.3 million dollars annually to prevent the invasion of a single hypothetical lake with a power plant. This figure can be compared with the 0.8 million dollars that the US Fish and Wildlife Service spent on all aquatic invaders in all US lakes in 2001 (Leung et al., 2002).

The costs of prevention comprise, for instance, costs of surveillance, labelling, import restrictions, compensation payments and post-monitoring. Often the benefits of not having the pest around outweigh these costs, but this is by no means inevitable (Mumford, 2002). In such a case, continued efforts to prevent the hazard from invading consume the limited resources and may possibly lead to other, more dangerous, hazards not being targeted with sufficient resources. Several European countries have, for instance, voluntarily renounced their protection systems regarding specific species (Heikkilä, 2006). On at least one occasion cost-benefit analysis has been in favour of denouncing a preventive policy (MacLeod et al., 2005).

Elbakidze and McCarl (2005) show that optimal pre-event detection depends on likelihood of disease introduction, spread rate, relative costs, ancillary benefits and effectiveness of mitigation. For slowly spreading diseases, investment in pre-outbreak activities is optimal only for very high outbreak probabilities. A somewhat similar conclusion is reached by Burnett et al. (2008), who suggest that it would be worthwhile spending money on finding small populations rather than to attempt to prevent all future introductions. Of course, investments in prevention and adaptation are not necessarily mutually exhaustive. For instance, Kim et al. (2005) show that prevention and control are complementary when the population size is small, but become substitutes as the population size increases.

In addition to biological factors, relative costs and benefits, and effectiveness of mitigation there are three further factors affecting the strategy choice between prevention and adaptation. The first of these is uncertainty. Mahul and Gohin (1999) note that if there is little uncertainty it may be worthwhile to wait and see until uncertainty is resolved. Thus, in practice adaptation may be preferred over prevention because of uncertainty. The relationship between resources invested in prevention and benefits thus acquired is very uncertain. Risk in this context has two dimensions: (1) the probability and level of damage associated with the biological hazard; and (2) the opportunity cost of resources spent on prevention of the biological hazard, which may or may not make an entry attempt. Finnoff et al. (2007) show that a risk-averse (i.e. precautionary) manager is likely to invest less resources in prevention and more in adaptation because the return for the investment in prevention is less certain. Hence adaptation may appear as the less risky management strategy, even though it is often also the more expensive one (Finnoff et al., 2007, see also Shogren, 2000 and Perrings, 2005).

The second additional factor is human adaptation, which is important to take into account when considering the merits of adaptation. In a relatively early theoretical analysis, Butler and Maher (1986) argue that by not taking the actions by the victims of an externality into account, the society may end up devoting too many resources to prevention. Similarly, Margolis et al. (2005) show that unaccounted for private actions result in the regulator choosing a level of prevention (tariff level) that is larger than is socially optimal. Shogren (2000) argues that once we acknowledge that people do adapt, assuming otherwise may lead to biased results.

Finally, the chosen objective of strategy optimisation may affect the desirability between prevention and adaptation. Barrett and Segerson (1997) note that besides Pareto efficiency there are other objectives that policies may seek, including minimising damage subject to a budget constraint or minimising expenditure subject to a given damage level. They show that under these constrained objectives some factors affecting the relative desirability of prevention versus adaptation may affect the decision differently from how they function under Pareto efficiency. For instance, in contrast to Pareto efficiency, under a budget constraint a reduction in uncertainty about the effectiveness of adaptation may lead to a decrease in the level of prevention and an increase in the level of adaptation. Adaptation may also be preferred if the government pays for prevention and has a strict budget constraint. Therefore, if public money can be more productively invested in preventing entry by Hazard B, it may be that Hazard A is best controlled through adaptation paid for by producers, although prevention might also in this case be a socially optimal strategy.

Lichtenberg and Penn (2003) - albeit in relation to chemical hazards - argue that prevention is not always the most cost-efficient strategy in the case of agricultural pollution. This is so when there are multiple sources of emissions, multiple sites affected and a widely adopted precautionary approach to uncertainty. More generally, based on the above discussion it might be argued that adaptation is relatively more attractive when: (1) the invasion probability is relatively low and the probable damage modest; (2) the effectiveness of adaptation is high; (3) adaptation strategies and incentives of individuals are taken into account; and (4) the decision-maker dislikes uncertainty regarding the outcome of the policies. Furthermore, as Waage and Mumford (2008) suggest, perhaps a greater focus should be put on building resilience to invasion (through, for instance, developing pest and disease resistance and vaccines as well as through diversification of production) rather than building walls around the protected systems.

#### 3.4.2 Category 3b: Cost-Efficiency of Management Methods

Cost-efficiency of different management methods within strategies is the next category reviewed here. Here, the management strategy (and perhaps the method) has already been agreed upon, but the question is how to utilise it most efficiently. This includes questions such as what is the economically optimal eradication radius in a disease outbreak, when and to what extent should vaccination be used, how much pesticide should be used to control a plant pest, and so forth. Separate questions also include how the costeffectiveness of the management method is affected by, for instance, resistance to chemical control substances and antibiotics or changes in the natural or production environment.

Studies in this category are numerous. To provide some examples, there have been studies on cost-effectiveness of alternative prevention measures (De Vos et al., 2005), stamping out and vaccination alternatives (Schoenbaum and Disney, 2003; Mangen et al., 2002; Mangen et al., 2001; Saatkamp et al., 2000), surveillance methods (Prattley et al., 2007; Powell et al., 2008; Saatkamp et al., 1997; Klinkenberg et al., 2005) and different food safety procedures (Jensen et al., 1998; Goldbach and Alban, 2006). As there are many studies in this category, various methods and study targets exist. The methods utilised include, for instance, stochastic dynamic programming (Houben et al., 1994), stochastic state transition models (van der Gaag et al., 2004; Schoenbaum and Disney, 2003), portfolio theory (Prattley et al., 2007), the scenario tree model (De Vos et al., 2005), the gravity model (Otsuki et al., 2001) and generalised linear mixed models (Chriél et al., 1999).

As an example of category 3b research methodology, consider optimal control. Using a traditional cost-benefit analysis it is possible to analyse a set of possible policies; for instance, the cost-efficiency of using a depopulation ring of 1, 3 or 5 kilometres, but it is not possible to analyse what would be the optimal radius of the depopulation ring (Kobayashi et al., 2007). To answer this, some sort of optimisation method is needed. Optimal control is a traditional method in resource economics and it is therefore not surprising that there are several applications to biosecurity. The application may be directly to optimal pest or disease control (Dehnen-Schmutz et al., 2004; Christiaans et al., 2007) or to, for instance, optimal feeding policy under animal movement restrictions (Niemi and Pietola, 2004). The control strategies have also been studied by Hilje and Stansly (2008), Johansson et al. (2005); Buhle et al. (2005), and Ross and Pollett (2007). The optimal control policy has been found to depend on, for instance, the available annual budget (Taylor and Hastings, 2004), policy- and sectorspecific factors (Breukers, 2007) and on the production function (Christiaans et al., 2007). It has also been found that the cost-effectiveness is reduced if all firms do not adopt the control procedure (van der Gaag et al., 2004) and that the adopted framework (for instance, expected utility) impacts on the optimal policy (Horan et al., 2002).

The study objects range from individual invasive species such as koalas (Ross and Pollett, 2007),

Bemisia tabaci (Hilje and Stansly, 2008) and greenhouse whitefly (McKee et al., 2008) to marine invasive species (Fernandez, 2008; Jones and Corona, 2008), ornamental garden plants (Dehnen-Schmutz et al., 2004), and individual species in specific countries (Blignaut et al., 2007). In the context of animal diseases the applications vary from individual diseases such as classical swine fever (De Vos et al., 2005) or hypothetical foot and mouth disease (Schoenbaum and Disney, 2003) to assessments of management systems for multiple diseases such as the pig identification and recording system in Belgium (Saatkamp et al., 1997) or hog producer income under contagious animal disease quarantine (Niemi and Pietola, 2004). In the context of zoonotic diseases and human health the applications include, for instance, salmonella (Goldbach and Alban, 2006; Chriél et al., 1999; van der Gaag et al., 2004; Jensen et al., 1998) and aflatoxin regulation (Otsuki et al., 2001).

Similarly to studies on the choice of management strategy, here a convergence between privately and socially optimal methods has also been found in several studies. Hilje and Stansly (2008) find that the method that is overall most economic (plastic cover) was also too expensive for the resource-poor farmers. Thus, although, for instance, Goldbach and Alban (2006) argue that it is the relative profitabilities of alternatives rather than the absolute values that are important, they are of little relevance if the private agents have no incentives to pay for either method. De Vos et al. (2005) point out that in determining the optimal method, one should account for the cost effectiveness but also for: (1) ease of implementation (small or large investments, are significant adaptations required); (2) allocation of costs and benefits (some measures may be too costly for private producers); (3) the cost-benefit ratio; and (4) attributable costs (beneficial side effects, e.g. reduce the probability of other hazards).

#### 3.4.3 Category 3c: Choice of Policy Instruments

In addition to considering the preferred control strategy (e.g. prevention) and preferred level of intervention, we are also interested in how that control strategy should be achieved. Should we, for instance, use command and control instruments to ban certain types of imports altogether, or should we perhaps tax the imports in a risk-based manner, or maybe we should resort to delivering information on the risks to all parties involved. Naturally, some policy instruments are complementary (e.g. information), whereas others can be seen as substitutes (e.g. trade ban and import tariffs cannot apply to the same products at the same time).

Economic analysis and development of policy instruments is one of the most promising areas in which economics can contribute to better biosecurity. For instance, Jones and Corona (2008) show that in the context of aquatic invasive species an ambient tax induces both short-run and long-run efficiency with minimal information requirements, and Blignaut et al. (2007) discuss the structure, size and distribution of a charge to eradicate invasive plants in South Africa over the next 25 years. Dehnen-Schmutz et al. (2004) show that a subsidy on control costs is the most effective and reliable method, and that increases in direct grants or in control volunteer numbers are less effective. In contrast, Hennessy (2007) finds that losses are smaller when production is concentrated, but that subsidies to small producers may exacerbate overall losses.

Overall, literature on this theme has been expanding vastly over the past few years, but is still relatively thinly dispersed. Topics that have recently been covered include the polluter pays approach (Jenkins, 2001), ambient taxes (Jones and Corona, 2008), user charges (Blignaut et al., 2007), tradable risk permits (Horan and Lupi, 2005), control subsidies (Dehnen-Schmutz et al., 2004; Hennessy, 2007) and Pigovian taxes (Knowler and Barbier, 2005), as well as discussion on economic incentives in general (Fernandez, 2008). Gren (2008) provides a discussion on policies available for the control of invasive species.

#### 3.4.4 Example: Biological Hazards and Import of Goods

The economics of biological hazards in relation to import of goods has been a topic of much research (e.g. Anderson et al., 2001; Oude Lansink, 2008) so we use it as an example of the types of studies conducted. When a particular good is proposed for import, there can be a screening of whether the good should be allowed for import or not. There has been some discussion on whether we would be better off ignoring the advice of the screening process or not (Smith et al., 1999; Caley et al., 2006). An empirical application of a similar question is provided by Keller et al. (2007),

who assess when risk assessment is worth it compared to letting all species in. They apply their model to the Australian ornamental industry, assuming that the decision to import a species is irreversible. Over relatively short periods there is not much difference in the net present value of the two policies (screen or do not screen), but for longer time periods screening creates large positive values and the fixed costs of undertaking the risk assessments have a negligible impact on the results. The authors point out that since the World Trade Organization requires that SPS-based risk reduction strategies must produce net economic gains, their model can be used to demonstrate that.

Import risk assessments (IRAs) can be used to evaluate whether some imports should be prevented altogether in order to protect the society from the risk of imported diseases and pests. Examples are provided by studies on the risks posed by import of animals vaccinated against foot and mouth disease and their products (Sutmoller and Casas Olascoaga, 2003), import of marker-vaccinated animals (Breidenbach et al., 2007), and the risk of BSE through imported animals and their products in Japan (Sugiura et al., 2003). Adamson and Cook (2007) point out that even in Australia, where relatively many import risk assessments have been undertaken, economic assessments are often limited to either estimating (1) how much consumers would benefit if free trade was allowed or (2) the economic impacts if a given pest or disease reaches the country. They argue that little work has been done to combine these two, which would be important in order to obtain a truthful estimate of the total impacts.

For instance, Cook (2008) studies the case where Chilean table grape producers were granted access to Australian markets, but not to Western Australia, which is free of many pests present elsewhere in the country. The outcome of the study is that the mean expected production damage from also allowing access to Western Australia would be around AU\$10.3 million, whereas consumer benefits would be only AU\$1.5 million. Hence there is no reason to allow market access. However, preventing trade in exotic species that also have beneficial impacts is not necessarily economically desirable. Knowler and Barbier (2005) studied saltcedar (Tamarisk spp.) in the United States, and their results indicate that to achieve the social optimum, the mere presence of risk associated with imports does not warrant prevention of commercial sales of exotic plant species with beneficial effects for consumers. Similarly, Otsuki et al. (2001) suggest that less stringent regulation may increase trade flows and income. Instead of bans, Knowler and Barbier (2005) suggest the use of market instruments, including a Pigovian tax, to regulate the industry and protect the society from the associated risks. Prevention of entry and the associated trade instruments are also studied by Horan et al. (2002), Costello and McAusland (2003), McAusland and Costello (2004), Horan and Lupi (2005), and Margolis et al. (2005).

In addition to being a question of choosing the right policy instrument, import risk assessments can be seen as a special case of hazard ranking (Sect. 3.3), because although not necessarily conducted in order to compare different biological hazards, the procedure is often so standardised that it allows the comparison of hazards for which an assessment has been conducted. This is especially so if the IRA follows consistent methodologies. In many cases IRAs are based on established risk assessment procedures; for instance, the OIE import risk analysis framework or the EPPO guidelines, which are accepted by the World Trade Organization as a justification for actions that could also be considered artificial barriers to trade. Naturally, even if the general framework is standard, the different economic components vary within the studies (e.g. which valuation methods have been used and what effects have been taken into account). Hence the different studies may still not be entirely comparable.

For goods and shipments accepted for import, regulation may still be necessary through, for instance, inspections or tariffs, which are meant to reduce the probability of harmful organisms entering the country. However, the inspection services are under pressure from the increasing import quantities as well as in many cases due to resource constraints. Further complications arise from the diverse range of imported goods and their origins. For instance, Costello et al. (2007) show that the risk varies by trading partner and that the cumulative number of introductions from a region is a concave function of imports. As for many countries it is vital to maintain an appropriate level of protection, a typical question in economic studies is how to establish efficient border protection measures.

The basic premise for many of the studies on optimal inspection strategies is that there is a number of shipments coming to a port, the probability that a crate within the shipment is infested is unknown, and the inspector has to decide how to allocate the scarce surveillance resources; for instance, how many crates to inspect and in which shipments, and how long a time should be used for the inspection. Research along these lines has been conducted by, for instance, Moffitt et al. (2008), Surkov et al. (2008), Batabyal and Beladi (2006), and Batabyal and Nijkamp (2006). An empirical assessment is provided by Work et al. (2005), who evaluate the effectiveness of monitoring the arrival of insect species in the United States through four cargo pathways.

As for tariffs, Paarlberg and Lee (1998) discuss the link between import tariffs and the level of health risk from imports with special reference to the foot and mouth disease. They show that the optimal level of tariffs is very sensitive to the risk of importing a contaminated product, as well as to the expected spread rate of the disease. More specifically, the infection rate of the imported goods has been shown to increase the optimal tariff level (McAusland and Costello, 2004). However, at very high levels of infection the optimal level of inspections decreases, possibly all the way to the point of no inspections. This occurs when most incoming goods are infected, and it is better to let them in without any inspections and instead charge a high tariff equal to the expected damage (McAusland and Costello, 2004).

However, too high a tariff is also socially suboptimal. Margolis et al. (2005) show that private actions result in the regulator choosing a higher tariff level than is socially optimal. Further, tariff escalation (the more processed the imported good, the higher the charged tariff) has been shown to increase the probability of hazard introduction (Tu et al., 2008). This is because it results in trade being biased towards primary commodities (with higher probability of harmful organisms) as opposed to processed products. Tu et al. (2008) show that a reduction in tariff escalation would increase allocative efficiency and reduce damage from invasive species.

In a case of unlimited research resources, the studies would consider the questions regarding whether to import or not, how to inspect and place tariffs optimally, how to control the species and disease if it arrived and escaped inspections, and so on, and then compare all possible combinations with each other to determine the optimal course of actions in the face of uncertainty. However, this is not practical in reality, and hence the questions are divided into smaller manageable questions. It is, as always, up to the researchers to place the research in the wider framework.

# 3.5 Category 4: Cost Estimates of Disease Outbreaks and Species Invasions

The final category discussed here includes the evaluations of individual invasion events or policies. There are hundreds of estimates of the damage caused by individual species or diseases, for both real invasions as well as for simulated ones. Some of these are related to specific countries or groups of countries, some to specific groups of species, and some to specific pests and diseases. They may be based on materialised damage from real invasions or on simulation of potential damage, and incorporate various and varying cost elements, making any comparison between the studies laborious or impossible. Some examples are provided in Table 1.

Figure 3 presents a collection of published empirical economic biosecurity studies by type and target country. Only countries for which more than one study was readily available are presented in the figure. The results were obtained with an internet search, as well as through the author's personal collection of papers. Only studies in which there is an empirical application to a real-life situation were included in the sample. Altogether, there are 231 studies in the sample. Although the literature search has not been exhaustive, general trends are easy to observe. The publications are divided such that 39% deal primarily with animal health, 25% with plant health, 11% with human health (only zoonotic diseases and biological food safety included in the sample) and 25% with environmental health (invasive alien species detrimental to the environment in general). Of course, sometimes it is difficult to precisely differentiate between plant health and environmental health.

The United States is the country in which most applications take place (35% of all studies in the sample), followed by Australia (13%), the United Kingdom (10%) and the Netherlands (9%). Together these four countries represent two-thirds of all studies. However, the domination of the US, European Union and Australia in these figures is not surprising considering that they are all industrialised countries who produce and export large quantities of agricultural products.

It also appears that many assessments in the United States (44% of all US studies) and South Africa (100%) relate to invasive species that pose a threat particularly to the environment. In contrast, in Australia (46% of Australian studies) the emphasis is

Level	Target	Annual cost estimate	Reference
Multi-country multi-hazard	Invasive alien species in 6 countries (USA, UK, Australia, South Africa, India and Brazil)	USD 314 billion	Pimentel et al., 2001
Single country	Invasive alien species in US	USD 120 billion	Pimentel et al., 1999, 2005
multi-hazard	Invasive alien species in China	USD 14.5 billion of which 83.4% indirect. Equivalent to 1.4% of Chinese GDP	Xu et al., 2006
	Invasive alien species in Canada	CAD 13–34 billion	Colautti et al., 2006
	Invasive alien species in Germany	100–265 million euros	Gebhardt, 1996; Reinhardt et al., 2003
	Weeds in Australia	AUD 4 billion	Martin, 2003
	35 animal diseases in UK	Economically most important are mastitis for cattle (£179.7 million), enzootic abortion for sheep (£23.8 million), swine influenza for pigs (£7.7 million) and salmonellosis for poultry (£104.7 million)	Bennett and Ijpelaar, 2003
	34 animal diseases in UK	£372-1061 million annually	Bennett and Ijpelaar, 2005
	Foodborne bacteria in US	USD 3.5–4.8 billion	Roberts, 1989
	Invasive pests in US	Quarter of agricultural gross national product	Schmitz and Simberloff, 1997; U.S. Congress, OTA, 1993
	South Africa fynbos	Value from R19 million under low valuation and poor management to R300 million under high valuation and good management. Cost of clearing alien plants 0.6–5% of the ecosystem value	Higgins et al., 1997
Multi-country single hazard	H5N1 avian influenza on international markets	First year market loss USD 1.4 billion to 14 billion, depending on scenario	FAO/CTD, 2006; Food Outlook, 2006
-	Asian longhorn beetle in Europe	90% of wood in infested areas could lose half of its value	MacLeod et al., 2002
	Salmonellosis in England and Sweden	For cost of illness (control cost) 2.6 pence (2.0 pence) per chicken in England and Wales and 0.5 pence (10.8 pence) per chicken in Sweden	Persson and Jendteg, 1992
	EU aflatoxin standard in 9 African countries	USD 670 million	Otsuki et al., 2001
	Foot and mouth disease in South America	Estimates benefits from various disease control strategies	Rich and Winter-Nelson, 2007
	<i>Mnemiopsis leidyi</i> in the Black Sea	Decline in profit from USD 17 million to USD 0.3 million	Knowler and Barbier, 2000; Knowler, 2005
	HPAI avian influenza globally	Simulate the impact of supply reductions on export prices using 16 scenarios and assuming constant demand, but do not calculate total monetary impacts	Djunaidi and Djunaidi, 2007
Single country single hazard	Hundreds, 231 depicted in Fig. 3		

 Table 1 Examples of individual invasion cost estimates

somewhat more on plant health and in the Netherlands (80%) and Denmark (67%) on animal health, which is not a big surprise, given the relative importance of an-

imal production in these two countries. In the United Kingdom, plant health constitutes about 39% and animal health about 48% of all studies.



Fig. 3 Reviewed biosecurity studies by category and country of application

These trends are similar to those found in a metaanalysis of the ecological invasion literature. Pysek et al. (2008) analysed 2670 papers dealing with 892 invasive species, and showed that all major groups are well studied, but most information on invasion mechanisms is based on a limited number of invaders. They also show a strong geographical bias, with Africa and Asia understudied and America and Europe having the highest number of both species studied and studies published. Almost half of all the invasive species and more than half of all studies analysed by Pysek et al. (2008) relate to North America. They also note that the position of Africa is largely determined by South Africa, representing two-thirds of research on the continent. The fact that these trends reflect on the economic literature is understandable, given that empirical applications in biosecurity economics are to a large extent dependent on availability of biological data or expertise.

Pysek et al. (2008) also point out that only 49 species were subject to 10 studies or more, the most studied being zebra mussel and Argentine ant. A similar conclusion of most research concentrating only on a few species was reached in the context of plant health by the EUPHRESCO project in the European Union (Buzy et al., 2007).

The publications were also classified by journal and year (Figs. 4 and 5). Figure 4 demonstrates that for animal health economics Preventive Veterinary Medicine is the primary journal of publication. On the other hand, for invasive species economics Ecological Economics is the top choice, followed by the Journal of Environmental Management and Biological Invasions. For plant health economics the most popular choice is Crop Protection, again followed by the Journal of Environmental Management and Agricultural Economics. It is also remarkable how thinly the studies are distributed over a wide variety of journals. In the sample of 231 studies there were 48 journals that contributed one paper each.

Just over 20% of the papers were published in economic journals, the major economic journals being Ecological Economics (14 studies), the American Journal of Agricultural Economics (10), Agricultural Economics (3) and Environmental and Resource Economics, the Journal of Agricultural and Applied Economics and the European Review of Agricultural Economics (2 each). Some 10% were in primarily natural science journals, the main journals being Ecological Modelling (3 studies), and Vaccine, Veterinary Microbiology, Ecological Applications and the International Journal of Food Microbiology (2 each). This



Fig. 4 Reviewed biosecurity studies by category and journal

also means that most papers were published in fairly interdisciplinary journals, as depicted in Fig. 4.

Figure 5 demonstrates that the volume of papers in the sample has increased gradually over the years and the field is growing steadily. It also seems that the number of papers on animal health economics has remained relatively stable over the past 15 years or so, while the number of papers on economics of invasive species has increased rapidly.

## 4 Discussion

It seems apparent that the broader the context (categories 1 and 2), the less economic research there is. Category 1 is virtually not studied at all, and category 2 is studied very little, with the exception of the risk ranking studies. These, however, would be likely to benefit from more rigorous economic considerations. In category 3 there are numerous studies concentrating on specific questions and employing a wide but relatively thin range of methodologies (see, e.g., Oude Lansink, 2008; Perrings et al., 2000). In category 4, a substantially greater number of studies exists, but these would benefit from some standardisation of approaches. The fact that in many assessments there are no clear common methods or frameworks (applying to both the methodologies used as well as to the types of costs included), cross-study comparison is relatively difficult. Yet, such comparison would be needed when deciding on how to allocate the available resources of the society. Of course, as mentioned earlier, the categories identified here are separated sections from the



Fig. 5 Reviewed biosecurity studies by category and year of publication

larger framework, which is seldom evaluated in individual papers, as demonstrated by the example of studies on import of goods.

It has been pointed out that two major contributions of economics to biosecurity discussion are, on one hand, to provide estimates on the impacts of invasions (category 4 in this paper) and hence improve the cost effectiveness and efficiency of control (category 3b), and on the other hand, to develop economic sanitary and phytosanitary measures (category 3c) (Evans et al., 2002). Shortle (2007) emphasises the importance of economic research to the development of effective and efficient management strategies (category 3a) as well as to understanding the co-evolution of economic and ecological systems. It is worthwhile to emphasise the role of economics in resource allocation and its use to determine how to prioritise different hazards, their management strategies, methods and policies.

Environmental and health impacts are notoriously difficult to value. However, as pointed out by Adamson and Cook (2007), even quantification of the area of impact and a list of native species at risk (and the level of risk) would be informative. It has also been suggested that the value of modelling is the actual process of working through the problems and issues rather than the identification of a final solution (Rushton and Upton, 2006). Modelling helps in conceptualising and thinking through complex problems analytically. In principle, for valuation of the effects of invasions or biosecurity policies almost any valuation method available in the economic literature is available. However, given the fact that we are dealing with thousands or tens of thousands of individual biological hazards, it seems clear that no nation has enough resources to undertake thorough studies of them all. Hence development of general assessment protocols, frameworks and prioritisation schemes for assessing the risk associated with biological hazards seems like a good candidate for a top priority research area.

The distribution of costs and benefits is an important issue, which is already beginning to affect the current animal health policies in the European Union. Also, in other contexts the distribution of costs and benefits has been found to be very uneven, as discussed in Section 3.4. Yet having a clearer idea about this would help in understanding the private incentives and subsequently in designing appropriate policies to account for the convergence of social and private interests. As Hennessy (2007) notes, designing public policies in biosecurity requires a clear understanding of the failure in private incentives.

Another challenge, as highlighted by Gramig and Wolf (2007), is that many studies are done on a disease-by-disease basis, yet preventative actions often

protect against multiple diseases. As they point out, preventative actions against multiple hazards can be thought of as either multi-product outputs of individual management practices, or as either positive or negative input externalities. Given that prevention is generally acknowledged to be a cost-effective strategy, it is somewhat surprising how few assessments there are of the ongoing costs and cost-efficiency of prevention at the farm level. The government-level costs are often known (although not necessarily published), but we often know very little about the farm-level costs and their impact on the level of prevention (Siekkinen et al., 2008). This is important, because these costs materialise every year, regardless of whether there is a pest or disease outbreak or not.

To conclude, we suggest the following top five priority areas (in no particular order) for study: (1) development of sophisticated economic policy instruments to mitigate biological hazards; (2) design of rigorous economic prioritisation frameworks to guide both academic research and government policies; (3) clearer understanding of the distribution of biosecurity impacts and the associated private incentives; (4) establishing a closer link between the costs and methods of prevention, and the economic risk; and (5) the role of human behaviour in biosecurity and how that behaviour can best be directed. We might also wish to find out to what extent lessons learned in some fields of biosecurity are applicable for other hazards, areas or circumstances. In general, it seems warranted to search for holistic approaches for instance, determining simultaneously the optimal strategy, optimal management methods and optimal policies to achieve the targets - and doing so for the right hazards. No small task.

## References

- Adamson D., Cook D. (2007) Re-examining economic options for import risk assessments, Murray Darling Program Working Paper M07#3, The University of Queensland, Australia.
- Andersson H., Lexmon Å., Robertsson J.-Å., Lundeheim N., Wierup M. (1997) Agricultural policy and social returns to eradication programs: the case of Aujeszky's disease in Sweden, Prev. Vet. Med. 29, 311–328.
- Anderson K., McRae C., Wilson D. (2001) The economics of quarantine and the SPS agreement, Centre for International Economic Studies and AFFA Biosecurity Australia, Adeleide and Canberra.

- Barrett J., Segerson K. (1997) Prevention and treatment in environmental policy design, J. Environ. Econ. Manager. 33, 196–213.
- Batabyal A.A., Beladi H. (2006) International trade and biological invasions: a queueing theoretic analysis of the prevention problem, Eur. J. Oper. Res. 170, 758–770.
- Batabyal A.A., Nijkamp P. (2006) Some probabilistic attributes of inspection policies in alien species management, Ecol. Econ. 60, 5–8.
- Bennett R., Ijpelaar J. (2003) Economic assessment of livestock diseases in Great Britain, Final Report to the Department for Environment, Food and Rural Affairs, 19 September 2003.
- Bennett R., Ijpelaar J. (2005) Updated estimates of the costs associated with thirty four endemic livestock diseases in Great Britain: A Note, J. Dairy Res. 66, 455–459.
- Berentsen P.B.M., Dijkhuizen A.A., Oskam A.J. (1992) A dynamic model for cost-benefit analyses of foot-and-mouth disease control strategies, Prev. Vet. Med. 12, 229–243.
- Bermejo M., Rodríquez-Teijeiro J.D., Illera G., Barroso A., Vilà C., Walsh P.D. (2006) Ebola outbreak killed 5000 gorillas, Science 314, 1564.
- Biosecurity Council (2003) Tiakina Aotearoa. Protect New Zealand: the biosecurity strategy for New Zealand. Biosecurity Council. New Zealand, available online at http://www. biosecurity.govt.nz/files//bio-strategy/biostrategy.pdf.
- Blignaut J.N., Marais C., Turpie J.K. (2007) Determining a charge for the clearing of invasive alien plant species (IAPs) to augment water supply in South Africa, Water SA 33, 27–34.
- Born W., Rauschmayer F., Bräuer I. (2005) Economic evaluation of biological invasions – a survey, Ecol. Econ. 55, 321–336.
- Breidenbach M.J., Engels M., Griot C., Stärk K.D.C. (2007) Quantitative risk assessment regarding the import of markervaccinated animals, A poster presented at the Annual Conference of the Society for Veterinary Epidemiology and Preventive Medicine (SVEPM), Helsinki/Espoo, Finland. 28–30 March 2007.
- Breukers A. (2007) Bio-economic modelling of brown rot in the Dutch potato production chain, PhD thesis, Wageningen University, The Netherlands.
- Buhle E.R., Margolis M., Ruesink J.L. (2005) Bang for buck: cost-effective control of invasive species with different life histories, Ecol. Econ. 52, 355–366.
- Burnett K., Kaiser B., Roumasset J. (2007) Economic lessons from control efforts for an invasive species: *Miconia calvescens* in Hawaii, J. For. Econ. 13, 151–167.
- Burnett K., Kaiser B., Pifafi B.A., Roumasset J. (2006) Prevention, eradication, and containment of invasive species: illustrations from Hawaii, Agric. Res. Econ. Rev. 35, 63–77.
- Burnett K.M., D'Evelyn S., Kaiser B.A., Nantamanasikarn P., Roumasset J.A. (2008) Beyond the lamppost: optimal prevention and control of the brown tree snake in Hawaii, Ecol. Econ. 67, 66–74.
- Butler R.V., Maher M.D. (1986) The control of externalities: abatement vs. damage prevention, Southern Econ. J. 52, 1088–1102.
- Buzy M., Marzin H., Anthoine G., Moens M., Inmand A. (2007) EC Sixth Framework ERA-NET Project EUPHRESCO, Report on the Mapping and Analysis of national Phytosanitary (Quarantine/Regulated Plant Health) Research Programmes.

- Cacho O.J., Wise R.M., Hester S.M., Sinden J.A. (2008) Bioeconomic modeling for control of weeds in natural environments, Ecol. Econ. 65, 559–568.
- Caley P., Lonsdale W.M., Pheloung P.C. (2006) Quantifying Uncertainty in Predictions of Invasiveness, Biol. Invasions 8, 277–286.
- Carvalho F.P. (2006) Agriculture, pesticides, food security and food safety, Environ. Sci. Policy 9, 685–692.
- CBD (Convention on Biological Diversity) (1992) Convention on Biological Diversity, available online at http://www. biodiv.org/doc/legal/cbd-en.pdf.
- Cembali T., Folwell R.J., Wandschneider P., Eastwell K.C., Howell W.E. (2003) Economic implications of a virus prevention program in deciduous tree fruits in the US, Crop Prot. 22, 1149–1156.
- Chriél M., Stryhn H., Dauphin G. (1999) Generalised linear mixed models analysis of risk factors for contamination of Danish broiler flocks with *Salmonella typhimurium*, Prev. Vet. Med. 40, 1–17.
- Christiaans T., Eichner T., Pethig R. (2007) Optimal pest control in agriculture, J. Econ. Dynam. Control 31, 3965–3985.
- Colautti R.I., Bailey S.A., van Overdijk C.D.A., Amundsen K., MacIsaac H.J. (2006) Characterised and projected costs of nonindigenous species in Canada, Biol. Invasions 8, 45–59.
- Connelly N.A., O'Neill C.R. Jr., Knuth B.A., Brown T.L. (2007) Economic impacts of zebra mussels on drinking water treatment and electric power generation facilities, Environ. Manag. 40, 105–112.
- Cook D. (2005) The 'Paradox of Thrips': identifying a critical level of investment in pest exclusion activities in Western Australia, Aust. Agribusiness Rev. 13, paper 11.
- Cook D., Proctor W. (2007) Assessing the threat of exotic plant pests, Ecol. Econ. 63, 594–604.
- Cook D.C. (2008) Benefit cost analysis of an import access request, Food Policy 33, 277–285.
- Cook D.C., Thomas M.B., Cunningham S.A., Anderson D.L., De Barro P.J. (2007) Predicting the economic impact of an invasive species on an ecosystem service, Ecol. Appl. 17, 1832–1840.
- Costello C., McAusland C. (2003). Protectionism, trade, and measures of damage from exotic species introductions, Am. J. Agr. Econ. 85, 964–975.
- Costello C., Springborn M., McAusland C., Solow A. (2007) Unintended biological invasions: does risk vary by trading partner? J. Environ. Econ. Manage. 54, 262–276.
- Council of Europe (2003) European strategy on invasive alien species, Strasbourg 5 September, 2003, T-PVS (2003), 7.
- Cowan P.E. (1992) The eradication of introduced Australian brushtail possums, *Trichosurus vulpecula*, from Kapiti Island, a New Zealand nature reserve, Biol. Conserv. 61, 217–226.
- Daily G.C., Ehrlich P.R. (1996) Impacts of development and global change on the epidemiological environment, Environ. Dev. Econ. 1, 311–346.
- Dalmazzone S. (2000) Economic factors affecting vulnerability to biological invasions, in: Perrings C., Williamson M., Dalmazzone S. (Eds.), 2000. The economics of biological invasions, Edward Elgar, Cheltenham.

- De Vos C.J., Saatkamp H.W., Huirne R.B.M. (2005) Costeffectiveness of measures to prevent classical swine fever introduction into The Netherlands, Prev. Vet. Med. 70, 235–256.
- Dehnen-Schmutz K., Perrings C., Williamson M. (2004) Controlling *Rhododendron ponticum* in the British Isles: an economic analysis, J. Environ. Manage. 70, 323–332.
- Dehnen-Schmutz K., Touza J., Perrings C., Williamson M. (2007) A century of the ornamental plant trade and its impact on invasion success, Diversity and Distributions 13, 527–534.
- Delfino D., Simmons P.J. (2000) Infectious diseases as invasives in human population, in: Perrings C., Williamson M., Dalmazzone S. (Eds.), 2000. The economics of biological invasions, Edward Elgar, Cheltenham.
- Dixon B. (2007) A rotten end, The Lancet 7, 375.
- Djunaidi H., Djunaidi A.C.M. (2007) The economic impacts of avian influenza on world poultry trade and the U.S. poultry industry: a spatial equilibrium analysis, J. Agric. Appl. Econ. 39, 313–323.
- Eiswerth M.E., van Kooten G.C. (2002) Uncertainty, economics and the spread of an invasive plant species, Am. J. Agric. Econ. 84, 1317–1322.
- Elbakidze L. (2003) The economics of agricultural bio-security: an interpretive literature review, Texan A&M University Report.
- Elbakidze L., McCarl B.A. (2005) Ex ante vs. ex post bioterrorism mitigation: better safe than sorry? Paper prepared for presentation at the American Agricultural Economics Association Annual Meeting, Providence, Rhode Island, 24–27 July 2005.
- European Commission (2006) Evaluation of the Community animal health policy (CAHP) 1995–2004 and alternatives for the future, available at: http://ec.europa.eu/food/animal/diseases/ strategy/main\_report\_part1\_en.pdf.
- Evans E.A., Spreen T.H., Knapp J.L. (2002) Economic issues of invasive pests and diseases and food safety, International Agricultural Trade and Policy Center Monograph Series MGTC02-2, University of Florida.
- FAO (2001) Economic impacts of transboundary plant pests and animal diseases, in: FAO, The State of Food and Agriculture 2001, FAO, Rome, Italy.
- FAO (2003) Committee on Agriculture, Seventeenth session: Biosecurity in Food and Agriculture, Rome, 31 March–4 April 2003.
- FAO/CTD (2006) Impact of animal disease outbreaks on livestock markets: an FAO analysis, available at: http:// www.fao.org/es/esc/common/ecg/157/en/Paper\_Impact\_of\_ Animal\_Disease\_Outbreaks\_on\_Livestock\_Markets\_FAO. pdf.
- Fasina F.O., Meseko A.C., Joannis T.M., Shittu A.I., Ularamu H.G., Egbuji N.A., Sulaiman L.K., Onyekonwu N.O. (2007) Control versus no control: options for avian influenza H5N1 in Nigeria, Zoonoses and Public Health 54, 173–176.
- Fernandez L. (2008) NAFTA and member country strategies for maritime trade and marine invasive species, J. Environ. Manage. 89, 308–321.

- Finnoff D., Shogren J.F., Leung B., Lodge D. (2007) Take a risk: preferring prevention over control of biological invaders, Ecol. Econ. 62, 216–222.
- Food Outlook (2006) Impact of animal disease outbreaks on livestock markets: an FAO analysis, No. 2, December 2006.
- Gebhardt H. (1996) Ecological and economic consequences of introductions of exotic wildlife (birds and mammals) in Germany, Wildlife Biol. 2, 205–211.
- Genovesi P. (2005) Eradications of invasive alien species in Europe: a review, Biol. Invasions 7, 127–133.
- Goldbach S.G., Alban L. (2006) A cost-benefit analysis of Salmonella-control strategies in Danish pork production, Prev. Vet. Med. 77, 1–14.
- Gramig B.M., Wolf C.A. (2007) Estimating within-herd preventive spillovers in livestock disease management, Am. J. Agric. Econ. 89, 1219–1225.
- Gren I.-M. (2008) Economics of alien invasive species management – choices of targets and policies, Boreal Environ. Res. 13, 17–32.
- Gutrich J.J., VanGelder E., Loope L. (2007) Potential economic impact of introduction and spread of the red imported fire ant, *Solenopsis invicta*, in Hawaii, Environ. Sci. Policy 10, 685–696.
- Harris R. (2005) Invasive ant risk assessment: ranking of highrisk species, available at: http://www.landcareresearch.co.nz/ research/biocons/invertebrates/Ants/ant\_pest\_risk.asp.
- Hastings A., Hall R.J., Taylor C.M. (2006) A simple approach to optimal control of invasive species, Theor. Popul. Biol. 70, 431–435.
- Heal G., Kriström B. (2002) Uncertainty and climate change, Environ. Res. Econ. 22, 3–39.
- Heath S.E. (2006) Challenges and options for animal and public health services in the next two decades, Office International des Épizootics (OIE) Scientific and Technical Review 25, 403–419.
- Heikkilä J., A review of risk prioritisation schemes of pathogens, pests and weeds: principles and practices. Forthcoming in *Agricultural and Food Science*.
- Heikkilä J., Peltola J. (2007) Phytosanitary measures under uncertainty: a case study of Colorado potato beetle in Finland, in: Oude Lansink A.G.J.M. (Ed.), New approaches to the economics of plant health, Wageningen UR Frontis Series, Vol. 20, Springer.
- Heikkilä J. Economics of invasive alien species: pre-emptive versus reactive control, Licentiate Thesis, MTT Discussion Papers 9/2006.
- Hennessy D.A. (2007) Biosecurity and spread of an infectious animal disease, Am. J. Agric. Econ. 89, 1226–1231.
- Higgins S.I., Turpie J.K., Constanza R., Cowling R.M., Le Maitre D.C., Marais C., Midgley G.F. (1997) An ecological economic simulation model of mountain fynbos ecosystems: dynamics, valuation and management, Ecol. Econ. 22, 155–169.
- Hilje L., Stansly P.A. (2008) Living ground covers for management of *Bemisia tabaci* (Gennadius) (Homoptera: Aleyrodidae) and tomato yellow mottle virus (ToYMoV) in Costa Rica, Crop Prot. 27, 10–16.
- Horan R.D., Lupi F. (2005) Tradeable risk permits to prevent future introductions of invasive alien species into the Great Lakes, Ecol. Econ. 52, 289–304.

- Horan R.D., Perrings C., Lupi F., Bulte E.H. (2002) Biological pollution prevention strategies under ignorance: the case of invasive species, Am. J. Agric. Econ. 84, 1303–1310.
- Houben E.H.P., Huirne R.B.M., Dijkhuizen A.A., Kristensen A.R. (1994) Optimal replacement of mastitic cows determined by a hierarchic Markov process, J. Dairy Sci. 77, 2975–2993.
- Houe H. (2003) Economic impact of BVDV infection in dairies, Biologicals 31, 137–143.
- Hundley K. (2007) Power plant: Is Arundo donax the answer to our power problems? St. Petersburg Times, 11 February 2007, available at: http://www.sptimes.com/2007/02/11/ Business/Power\_plant\_Is\_Arun.shtml.
- IMF (2006) The global economic and financial impact of an avian flu pandemic and the role of the IMF, Avian Flu Working Group.
- James A.D., Rushton J. (2002) The economics of foot and mouth disease, Office International des Épizootics (OIE) Scientific and Technical Review 21, 637–644.
- Jenkins P. (2001) Who should pay? Economic dimensions of preventing harmful invasions through international trade and travel, in: McNeely J.A. (Ed.), 2001. The great reshuffling: human dimensions of invasive alien species, IUCN.
- Jensen H.H., Unnevehr L.J., Gómez M.I. (1998) Costs of improving food safety in the meat sector, J. Agric. Appl. Econ. 30, 83–94.
- Jensen R. (2002) Economic policy for invasive species, Working Paper, available online at http://www.nd.edu/~rjensen1/ workingpapers/InvasiveSpecies.pdf.
- Johansson R., Livingston M., Westra J., Guidry K. (2005) Preliminary estimates of U.S. economic and environmental effects of treating and adjusting to Asian soybean rust, Presentation at Northeasters Agricultural and Resource Economics Association's Economics of Invasive Species workshop, Annapolis, Maryland, 14–15 June 2005.
- Jones K.R., Corona J.P. (2008) An ambient tax approach to invasive species, Ecol. Econ. 64, 534–541.
- Keller R.P., Lodge D.M., Finnoff D.C. (2007) Risk assessment for invasive species produces net bioeconomic benefits, Proc. Nat. Acad. Sci. 104, 203–207.
- Kim C.S., Lubowski R., Lewandrowski J. (2005) Prevention or control? Optimal government policies for invasive species management, A paper presented at the Northeastern Agricultural and Resource Economics Association's (NAREA) workshop on Invasive Species, Annapolis, Maryland, 14–15 June 2005, available online at: http://www.arec.umd.edu/ llynch/personal%20files/NAREA-IS-Work.htm.
- Klinkenberg D., Nielen M., Mourits M.C.M., de Jong M.C.M. (2005) The effectiveness of classical swine fever surveillance programmes in The Netherlands, Prev. Vet. Med. 67, 19–37.
- Knowler D. (2005) Reassessing the costs of biological invasion: *Mnemiopsis leidyi* in the Black sea, Ecol. Econ. 52, 187–199.
- Knowler D., Barbier E.B. (2000) The economics of an invading species: a theoretical model and case study application, in: Perrings C., Williamson M., Dalmazzone S. (Eds.), 2000. The economics of biological invasions, Edward Elgar, Cheltenham.
- Knowler D., Barbier E. (2005) Importing exotic plants and the risk of invasion: are market-based instruments adequate? Ecol. Econ. 52, 341–354.

- Kobayashi M., Carpenter T.E., Dickey B.F., Howitt R.E. (2007) A dynamic, optimal disease control model for foot-andmouth disease: I. Model description, Prev. Vet. Med. 79, 257–273.
- Leung B., Lodge D.M., Finnoff D., Shogren J.F., Lewis M.A., Lamberti G. (2002) An ounce of prevention or a pound of cure: bioeconomic risk analysis of invasive species, Proc. R. Soc. Lond. B. 269, 2407–2413.
- Lichtenberg E., Penn T.M. (2003) Prevention versus treatment under precautionary regulation: a case study of groundwater contamination under uncertainty, Am. J. Agric. Econ. 85, 44–58.
- MacLeod A., Baker R.H.A., Cannon R.J.C. (2005) Costs and benefits of European Community (EC) measures against invasive alien species – current and future impacts of *Diabrotica virgifera virgifera* in England & Wales, A paper prepared for symposium on Introduction and Spread of Invasive Species, 9–11 June 2005, Berlin, Germany.
- MacLeod A., Evans H.F., Baker R.H.A. (2002) An analysis of pest risk from an Asian longhorn beetle (*Anoplophora* glabripennis) to hardwood trees in the European community, Crop Prot. 21, 635–645.
- Mahul O., Gohin A. (1999) Irreversible decision making in contagious animal disease control under uncertainty: an illustration using FMD in Brittany, Eur. Rev. Agric. Econ. 26, 39–58.
- Maijala R., Ranta J., Seuna E., Peltola J. (2005) The efficiency of the Finnish salmonella control programme, Food Control, 16, 669–675.
- Mangen M.-J.J., Burrell A.M. (2003) Who gains, who loses? Welfare effects of classical swine fever epidemics in the Netherlands, Eur. Rev. Agric. Econ. 30, 125–154.
- Mangen M.-J.J., Jalvingh A.W., Nielen M., Mourits M.C.M., Klinkenberg D., Dijkhuizen A.A. (2001) Spatial and stochastic simulation to compare two emergency-vaccination strategies with a marker vaccine in the 1997/8 Dutch classical swine fever epidemic, Prev. Vet. Med. 48, 177–200.
- Mangen M.-J.J., Nielen M., Burrell A.M. (2002) Simulated effect of pig-population density on epidemic size and choice of control strategy for classical swine fever epidemics in The Netherlands, Prev. Vet. Med. 56, 141–163.
- Margolis M., Shogren J.F., Fischer C. (2005) How trade policies affect invasive species control, Ecol. Econ. 52, 305–313.
- Martin P. (2003) Killing us softly Australia's green stalkers: a call to action on invasive plants, and a way forward, CRC for Australian Weed Management.
- McAusland C., Costello C. (2004) Avoiding invasives: traderelated policies for controlling unintentional exotic species introductions, J. Environ. Econ. Manage. 48, 954–977.
- McInerney J., Kooij D. (1997) Economic analysis of alternative AD control programmes, Vet. Microbiol. 55, 113–121.
- McInerney J.P., Howe K.S., Schepers J.A. (1992) A framework for the economic analysis of disease in farm livestock, Prev. Vet. Med. 13, 137–154.
- McKee G.J., Goodhue R.E., Zalom F.G., Carter C.A., Chalfant J.A. (2008) Population dynamics and the economics of invasive species management: the greenhouse whitefly in California-grown strawberries, J. Environ. Manage. (in press).

- McKenzie J., Simpson H., Langstaff I. (2007) Development of methodology to prioritise wildlife pathogens for surveillance, Prev. Vet. Med. 81, 194–210.
- Mehta S.V., Haight R.G., Homans F.R., Polasky S., Venette R.C. (2007) Optimal detection and control strategies for invasive species management, Ecol. Econ. 61, 237–245.
- Moffitt L.J., Osteen G.D. (2006) Prioritizing invasive species threats under uncertainty, Agric. Res. Econ. Rev. 35, 41–51.
- Moffitt L.J., Stranlund J.K., Field B.C., Osteen C.D. (2008) Robust inspection for invasive species with a limited budget, J. Environ. Manage. 89, 293–299.
- Mort M., Convery I., Baxter J., Bailey C. (2005) Psychosocial effects of the 2001 UK foot and mouth disease epidemic in a rural population: qualitative diary based study, BMJ, DOI: 10.1136/bmj.38603.375856.68 (published 7 October 2005).
- Mukhebi A.W., Chamboko T., O'Callaghan C.J., Peter T.F., Kruska R.L., Medley G.F., Mahan S.M., Perry B.D. (1999) An assessment of the economic impact of heartwater (*Cowdria ruminantium* infection) and its control in Zimbabwe, Prev. Vet. Med. 39, 173–189.
- Mumford J.D. (2002) Economic issues related to quarantine in international trade, Eur. Rev. Agric. Econ. 29, 329–348.
- Mumford J.D., Temple M.L., Quinlan M.M., Gladders P., Blood-Smyth J.A., Mourato S.M., Makuch Z., Crabb R.J. (2000) Economic policy evaluation of MAFF's Plant Health Programme. Report to Ministry of Agriculture Fisheries and Food, London, United Kingdom, available at: http://statistics. defra.gov.uk/esg/evaluation/planth/.
- Niemi J.K., Pietola K. (2004) Pigmeat producer incentives and moral hazard under contagious animal disease restrictions, Paper presented at 86th EAAE Seminar Farm income stabilisation: what role should public policy play? Anacapri.
- Niemi J.K., Lehtonen H., Pietola K., Lyytikäinen T., Raulo S. (2008) Simulated financial losses of classical swine fever epidemics in the Finnish pig production sector, Prev. Vet. Med. 84, 194–212.
- Olson L.J. (2006) The economics of terrestrial invasive species: a review of the literature, Am. J. Agric. Econ. 35, 178–194.
- Olson L.J., Roy S. (2002) The economics of controlling a stochastic biological invasion, Am. J. Agric. Econ. 84, 1311– 1316.
- Otsuki T., Wilson J.S., Sewadeh M. (2001) Saving two in a billion: a case study to quantify the trade effect of European food safety standards on African exports, Food Policy 26, 495–514.
- Oude Lansink A.G.J.M. (2007) New approaches to the economics of plant health, Wageningen UR Frontis Series, Vol. 20, Springer.
- Paarlberg P., Lee J. (1998) Import restrictions in the presence of a health risk: an illustration using FMD, Am. J. Agric. Econ. 80, 175–183
- Paarlberg P.L., Hillberg Seitzinger A., Lee J.G. (2007) Economic impacts of regionalization of a highly pathogenic avian influenza outbreak in the United States, J. Agric. Appl. Econ. 39, 325–333.
- Parker C., Caton B.P., Fowler L. (2007) Ranking nonindigenous weed species by their potential to invade the United States, Weed Sci. 55, 386–397.
- Peeler E.J., Murray A.G., Thebault A., Brun E., Giovaninni A., Thrush M.A. (2007) The application of risk analysis in aquatic animal health management, Prev. Vet. Med. 81, 3–20.

- Perrault A., Carroll Muffett W. (2001) Encouraging prevention, developing capacity and providing accountability: a strategy for addressing international invasive alien species issues, Center for International Environmental Law Discussion Paper, Washington DC.
- Perrings C. (2000) The economics of biological invasions, in: Preston G., Brown A.G., van Wyk E. (Eds.), 2000. Best Management Practices for Preventing and Controlling Invasive Alien Species, Symposium Proceedings, Cape Town, South Africa: Working for Water Programme.
- Perrings C. (2005) Mitigation and adaptation strategies for the control of biological invasions, Ecol. Econ. 52, 315–325.
- Perrings C., Williamson M., Dalmazzone S. (2000) The economics of biological invasions, Edward Elgar, Cheltenham.
- Perry B.D., Randolph T.F. (1999) Improving the assessment of the economic impacts of parasitic diseases and of their control in production animals, Vet. Parasitol. 84, 145–168.
- Persson U., Jendteg S. (1992) The economic impact of poultryborne salmonellosis: how much should be spent on prophylaxis? Int. J. Food Microbiol. 15, 207–213.
- Pifafi B.A., Rousmasset J.A. (2007) Some resource economics of invasive species, A paper presented at the 15th Annual Conference of the European Association of Environmental and Resource Economists (EAERE), 27–30 June 2007, Thessaloniki, Greece.
- Pimentel D., Lach L., Zuniga R., Morrison D. (1999) Environmental and economic costs associated with non-indigenous species in the United States, http://www.news.cornell.edu/ releases/Jan99/species\_costs.html.
- Pimentel D., McNair S., Janecka J., Wightman J., Simmonds C., O'Connell C., Wong E., Russel L., Zern J., Aquino T., Tsomondo T. (2001) Economic and environmental threats of alien plant, animal, and microbe invasions, Agric. Ecosyst. Environ. 84, 1–20.
- Pimentel D., Zuniga R., Morrison D. (2005) Update on the environmental and economic costs associated with alien-invasive species in the United States, Ecol. Econ. 52, 273–288.
- Powell M., Scott A., Ebel E. (2008) Analyzing BSE surveillance in low prevalence countries, Prev. Vet. Med. 83, 337–346.
- Prattley D.J., Morris R.S., Stevenson M.A., Thornton R. (2007) Application of portfolio theory to risk-based allocation of surveillance resources in animal populations, Prev. Vet. Med. 81, 56–69.
- Pysek P., Richardson D.M., Pergl J., Jarosik V., Sixtová Z., Weber E. (2008) Geographical and taxonomic biases in invasion ecology, Trends Ecol. Evol. 23, 237–244.
- Ranjan R., Marshall E., Shortle J. (2008) Optimal renewable resource management in the presence of endogenous risk of invasion, J. Environ. Manage. 89, 273–283.
- Reinhardt F., Herle M., Bastiansen F., Streit B. (2003) Economic impact of the spread of alien species in Germany, German Federal Environmental Agency report number 80/2003.
- Renn O. (2006) Risk governance: towards an integrative approach, White Paper No. 1, International Risk Governance Council, Geneva.
- Rich K.M., Winter-Nelson A. (2007) An integrated epidemiological-economic analysis of foot-and-mouth disease: applications to the Southern Cone of South America, Am. J. Agric. Econ. 89, 682–697.

- Roberts T. (1989) Human illness costs of foodborne bacteria, Am. J. Agric. Econ. 71, 468–474.
- Rosengren H., Heikkilä J. (2009) Risk-based classification of animal diseases: the case of poultry, A poster presented at the 12th International Symposium on Veterinary Epidemiology and Economics (ISVEE XII), 10–14 August 2009, Durban, South Africa.
- Ross J.V., Pollett P.K. (2007) On costs and decisions in population management, Ecol. Model. 201, 60–66.
- Rossides S.C. (2002) A farming perspective on the 2001 foot and mouth disease epidemic in the United Kingdom, Office International des Épizootics (OIE) Scientific and Technical Review 21, 831–838.
- Rushton J., Upton M. (2006) Investment in preventing and preparing for biological emergencies and disasters: Social and economic costs of disasters versus costs of surveillance and response preparedness, Office International des Épizootics (OIE) Scientific and Technical Review 25, 378–388.
- Russell J., Choquenot D., Clout M., Beggs J. (2006) Developing a framework for prioritising pest management policy, Report prepared for the Auckland Regional Council, Centre for Biodiversity and Biosecurity, Auckland.
- Saatkamp H.W., Berentsen P.B.M., Horst H.S. (2000) Economic aspects of the control of classical swine fever outbreaks in the European Union, Vet. Microbiol. 73, 221–237.
- Saatkamp H.W., Dijkhuizen A.A., Geers R., Huirne R.B.M., Noordhuizen J.P.T.M., Goedseels V. (1997) Economic evaluation of national identification and recording systems for pigs in Belgium, Prev. Vet. Med. 30, 121–135.
- Schmitz D.C., Simberloff D. (1997) Biological invasions: a growing threat, Issues in Science and Technology 13, 33–40.
- Schoenbaum M.A., Disney W.T. (2003) Modeling alternative mitigation strategies for a hypothetical outbreak of footand-mouth disease in the United States, Prev. Vet. Med. 58, 25–52.
- Schrader G., Unger J.-G. (2003) Plant quarantine as a measure against invasive alien species: the framework of the International Plant Protection Convention and the plant health regulations in the European Union, Biol. Invasions 5, 357–364.
- Sharov A.A. (2004) Bioeconomics of managing the spread of exotic pest species with barrier zones, Risk Analysis 24, 879– 892.
- Sharov A.A., Liebhold A.M. (1998) Bioeconomics of managing the spread of exotic pest species with barrier zones, Ecol. Appl. 8, 833–845.
- Shine C., Williams N., Gündling L. (2000) A guide to designing legal and institutional frameworks on alien invasive species, Environmental Policy and Law Paper No. 40, Environmental Law Centre, The World Conservation Union.
- Shogren J.F. (2000) Risk reduction strategies against the 'explosive invader', in: Perrings C., Williamson M., Dalmazzone S. (Eds.), 2000. The economics of biological invasions, Edward Elgar, Cheltenham, 249 p.
- Shortle J.S. (2007) Economic incentives for managing infectious animal disease: discussion, Am. J. Agric. Econ. 89, 1239–1240.
- Siekkinen K.-M., Heikkilä J., Tammiranta N., Rosengren H. (2008) The costs of biosecurity at the farm level: the case

of Finnish broiler, A poster presented at the 12th Congress of the European Association of Agricultural Economists (EAAE), 26–29 August 2008, Gent, Belgium.

- Simberloff D. (2003) How much information on population biology is needed to manage introduced species, Conserv. Biol. 17, 83–92.
- Smith C.S., Lonsdane W.M., Fortune J. (1999) When to ignore advice: invasion predictions and decision theory, Biol. Invasions, 1, 89–96.
- Sugiura K., Ito K., Yokoyama R., Kumagai S., Onodera T. (2003) A model to assess the risk of the introduction into Japan of the bovine spongiform encephalopathy agent through imported animals, meat and meat-and-bone meal, Office International des Épizootics (OIE), Scientific and Technical Review 22, 777–794.
- Surkov I.V., Oude Lansink A.G.J.M., van Kooten O., van der Werf W. (2008) A model of optimal import phytosanitary inspection under capacity constraint, Agric. Econ. 38, 363–373.
- Sutmoller P., Casas Olascoaga R. (2003) The risks posed by the importation of animals vaccinated against foot and mouth disease and products derived from vaccinated animals: a review, Office International des Épizootics (OIE) Scientific and Technical Review 22, 823–835.
- Taylor C.M., Hastings A. (2004) Finding optimal control strategies for invasive species: a density-structured model for *Spartina alterniflora*, J. Appl. Ecol. 41, 1049–1057.
- Tu A.T., Beghin J., Gozlan E. (2008) Tariff escalation and invasive species damages, Ecol. Econ. 67, 619–629.
- U.S. Congress, Office of Technology Assessment (1993) Harmful non-indigenous species in the United States, OTA-F-565, Washington DC, available online at http://www. wws.princeton.edu/cgi-bin/byteserv.prl/~ota/disk1/1993/ 9325/9325.pdf.
- Valeeva N.I., Huirne R.B.M., Meuwissen M.P.M., Oude Lansink A.G.J.M. (2007) Modeling farm-level strategies for improving food safety in the dairy chain, Agric. Syst. 94, 528–540.

- Valeeva N.I., Meuwissen M., Oude Lansink A., Huirne R. (2006) Cost implications of improving food safety in the Dutch dairy chain, Eur. Rev. Agric. Econ. 33, 511–541.
- van der Gaag M.A., Saatkamp H.W., Backus G.B.C., van Beek P., Huirne R.B.M. (2004) Cost-effectiveness of controlling Salmonella in the pork chain, Food Control 15, 173–180.
- Virtue J. (2007) Weed Risk Management in South Australia. Presentation at International Weed Risk Assessment Workshop (IWRAW) 2007, Perth (WA), Australia, 14–15 September 2007.
- Waage J.K., Mumford J.D. (2008) Agricultural biosecurity, Philos. Trans. R. Soc. Lond. B 363, 863–876.
- Waage J.K., Fraser R.W., Mumford J.D., Cook D.C., Wilby A. (2005) A new agenda for biosecurity, Faculty of Life Sciences, Imperial College London, London.
- Weber E., Gut D. (2004) Assessing the risk of potentially invasive plant species in central Europe, J. Nature Conserv. 12, 171–179.
- Weigle S.M., Smith L.D., Carlton J.T., Pederson J. (2005) Assessing the risk of introducing exotic species via the live marine species trade, Conserv. Biol. 19, 213–223.
- WHO (2007) Food safety and foodborne illness, Fact sheet Number 237, Reviewed March 2007.
- Wolfe N.D., Panosian Dunavan C., Diamond J. (2007) Origins of major human infectious diseases, Nature 447, 279–283.
- Wooldridge M., Hartnett E., Cox A., Seaman M. (2006) Quantitative risk assessment case study: smuggled meats as disease vectors, Office International des Épizootics (OIE) Scientific and Technical Review 25, 105–117.
- Work T.T., McGullough D.G., Cavey J.F., Komsa R. (2005) Arrival rate of nonindigenous insect species into the United States through foreign trade, Biol. Invasions 7, 323–332.
- Xu H., Ding H., Li M., Qiang S., Guo J., Han Z., Huang Z., Sun H., He S., Wu H., Wan F. (2006) The distribution and economic losses of alien species invasion to China, Biol. Invasions 8, 1495–1500.
- Zavaleta E. (2000) The economic value of controlling an invasive shrub, Ambio 29, 462–467.

# **Describing and Locating Cropping Systems on a Regional Scale**

Delphine Leenhardt, Frédérique Angevin, Anne Biarnès, Nathalie Colbach, and Catherine Mignolet

## Contents

1	Introduction		
2	Using the Concept of Cropping Systems on a Regional Scale		
3	Objectives of Describing and Locating Cropping Systems on a Regional Scale		
4	Scale Issues		
5	Describing Cropping Systems by Relevant Variables		
6	Collecting Data         6.1       Direct Collection         6.2       Indirect Collection: Use of Determining         Factors	90 90 91	
7	Conclusion	93	
Re	ferences	93	

D. Leenhardt (⊠)

Abstract At regional scale issues such as diffuse pollution, water scarcity and pollen transfer are closely related to the diversity and location of cropping systems because agriculture interacts with many other activities. Although sustainable land use solutions for territorial development and natural resource management are needed, very few agro-environmental studies account for both the coherence and the spatial variability of cropping systems. The originality of this article is to review methods that describe and locate cropping systems within large areas. We mainly based our analysis on four case studies using the concept of cropping systems on a regional scale, but differing in their objectives and extents. We found that describing and locating cropping systems in space meets not only decision-making stakes but also a scientific stake that allows multi-simulations over large areas when models require cropping system information. Simulation models are indeed necessary when the study aims at estimating cropping system externalities. Then, the involved process determines the extent, and the model determines the support unit, unless socio-economic considerations prevail. In this case, as well as when no model is involved, it is often considerations related to stakeholders that determine extent and support unit choices. On a regional scale, the cropping system must be described by only a few variables whose selection depends on the study objective and the involved processes. Collecting cropping system information for all support units is often simplified by identifying determining factors of cropping systems. However, obtaining deterministic relations between easily accessible factors and cropping system variables is not always possible, and sometime accessing modalities of determining factors for all support units is also difficult.

INRA, UMR1248 – AGIR (Agrosystèmes et développement territorial), BP 52627, 31326 Castanet Tolosan Cedex, France e-mail: Delphine.Burger-Leenhardt@toulouse.inra.fr

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_6,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Leenhardt et al., Agron. Sustain. Dev. 30 (2010) 131–138. DOI: 10.1051/agro:2009002

We found that describing and locating cropping systems relied very much on expertise and detailed survey data. The development of land management practice monitoring would facilitate this description work.

**Keywords** Agricultural practices • Spatial distribution • Modelling • Cropping system • Scale change • Environmental impact

## 1 Introduction

On a regional scale, where agriculture interacts with many other activities, and where various farmers and various cropping systems interact together, many management problems are closely related to the nature, diversity and location of cropping systems. For example, to adjust the quality of cereal lots to millers' requirements, the cooperative needs to know the distribution of the performance, e.g. yield and protein content, of the collected crop over its collect basin (Le Bail, 2005). This results from the characteristics of the cropping systems such as preceding crop, sowing dates, varieties and fertilisation. For agricultural planning of a country or to assist marketing decisions for farmers and grain traders, it is also necessary to predict crop or forage yield on a national or regional basis (e.g. Lal et al., 1993; Rosenthal et al., 1998; Chipanshi et al., 1999; Donet, 1999; Faivre et al., 2000; Yun, 2003). Calculating water requirements for agriculture within the area served by a water provider requires knowing where, in terms of soil and weather conditions, irrigated cropping systems are located, and to distinguish whether these cropping systems are intensively irrigated or not (e.g. Sousa and Santos Pereira, 1999; Heinemann et al., 2002; Leenhardt et al., 2004). Similarly, an accurate knowledge of cropping systems is needed to estimate nitrogen or pesticide pollution of streams or groundwater from agricultural land (Beaujouan et al., 2001; Gomez and Ledoux, 2001; Louchart et al., 2001; Biarnès et Colin, 2006) or to evaluate the risk of occurrence of geneticallymodified (GM) seeds in non-GM harvests (Colbach, 2008; Angevin et al., 2008). The existence of dispersal processes such as water runoff and pollen dispersal may require, in addition, knowing the exact and relative locations of cropping systems. It may also require knowing the location of semi-natural areas such

as grass strips and roadsides because a cropping system in one location can reduce or increase the effect of a neighbouring cropping system. For instance, herbicide runoff from a field may be reduced by a grass strip; GM propagation from a GM field will be increased if fields of similar GM cropping systems are surrounding it.

This rapid overview shows that information regarding crops and agricultural practices is required for solving many agri-environmental management problems. Accounting for the strong links that can exist between technical operations, between successive crops, and between crops and technical operations, can considerably improve decision-making by evoking more sustainable management decisions. Describing and locating cropping systems in space thus meets a public decision-making need.

This outlook article aims at presenting how cropping systems can be described and located within large and/or heterogeneous areas including a great number of fields, and therefore likely to present a great variability of cropping systems. For this purpose, it is based on four groups of studies covering a wide range of agri-environmental problems and extents (Table 1). They all use the concept of the cropping system but developed the necessary simplifications to account for the spatial variability of cropping systems on a regional scale. First, we present the concept of the cropping system and its classical use on a regional scale. Then, we present the objectives for which it is necessary to describe and locate cropping systems, which lead us to specify when environmental process models are used. In a second section, we highlight some scale issues. The third section states the problem of the description of cropping systems, in particular the choice of the variables to use. The collection of these variables at the required resolution is the subject of the last section, which distinguishes the direct collection of these variables from an indirect way that involves identifying and collecting determining factors of the cropping systems.

# 2 Using the Concept of Cropping Systems on a Regional Scale

The concept of the cropping system used in this article was developed in the framework of traditional French agronomic research. It was precisely defined

	Objective and thematic						
Case study	of the study	References	Extent	Support unit			
A	Effect of cropping systems on regional gene flow	Angevin et al., 2002 Colbach, 2008	Group of fields <10 km <sup>2</sup>	Field $1 \text{ m}^2 - <50 \text{ ha}$			
В	Impact of weed control practices on water quality in a small catchment	Biarnès and Colin, 2006 Biarnès et al., 2004 Louchart et al., 2001	Water catchment 75 km <sup>2</sup>	Sub-catchment area 1–5 km <sup>2</sup>			
С	Estimating regional irrigation demand	Leenhardt et al., 2004 Maton, 2006 Maton et al., 2007a, b Clavel and Leenhardt, 2008	Irrigated perimeter 500 km <sup>2</sup>	Not a priori defined (Maton et al., 2007a) or Intersection of a small agricultural region and a water management region ~100 km <sup>2</sup> (Clavel and Leenhardt, 2008)			
D	Simulating nitrate flow in the Seine basin	<i>Gomez and Ledoux, 2001</i> <i>Ledoux et al., 2007</i> Le Ber et al., 2006 Mignolet et al., 2001, 2004, 2007	River basin 95000 km <sup>2</sup>	Small agricultural region ~400 km <sup>2</sup>			

Table 1 Short description of the four case studies on which the analysis of the article is mainly based, with their references and, in italics, related references

by Sebillotte (1974), for an area managed homogeneously, as "the crops, their succession order and the crop management systems associated with each crop". The cropping system was thus considered as a subsystem of the complex land use system, its typical spatial scale being the field plot. On such a scale, Meynard et al. (2001) insisted on the temporal coherence of the cropping system: the crop management system corresponds to a logical and coordinated succession of actions; the various technical choices made within a given cropping system are not independent. Later, Veldkamp et al. (2001) showed that agronomic research could also be relevant at the landscape level, and, on such scale, Benoît and Papy (1998) and Jouve (2006) noticed that crop successions and crop management systems did not result from a random process but were the result of logic and decision processes. These authors highlighted the spatial coherence of cropping systems.

Nevertheless, the concept of the cropping system is rarely emphasised on scales greater than the field. The main reason is probably that information related to cropping systems is lacking or difficult to collect when a great number of fields is concerned. Therefore, the description of agricultural activities and land use is most often very simplified on such a "regional"<sup>1</sup> scale. Simplification often leads to neglecting the concept of the cropping system. A first simplification consists of restricting the concept of the cropping system to its primary characteristic, i.e. the knowledge of the cultivated species, as noted by Jouve (2006). This schematisation is common in the remote-sensing approaches used to describe land use data for environmental issues (Martin et al., 2006). The description of agricultural land cover is usually restricted to soil occupancy, e.g. cultivated areas, forests, grassland and built-up areas; and to main crop groups such as annual versus permanent crops and, at the most, crop species such as maize and wheat (Verburg and Veldkamp, 2001). Details on crop management systems and crop successions are usually omitted. A second level of simplification consists of describing the crop management system as a fixed and uniform sequence of technical operations. This sequence can either consist of average practices

<sup>&</sup>lt;sup>1</sup> We use here the word "region" to refer to any area so large or heterogeneous that it includes a great number of fields impossible to survey. In this article, a "region" could be a small area of less than 10 km<sup>2</sup> with many small fields, as well as a very big river catchment ( $\sim$ 100 000 km<sup>2</sup>).

(Moen et al., 1994) or correspond to recommended management guidelines (Yun, 2003). The spatial variability in crop management systems is often neglected and the same recommended or averaged practices are used for a large part or the whole spatial extent. The temporal coherence of the management system is also often limited: crop management systems are expressed as a fixed calendar of technical operations. This second level of schematisation is frequently adopted in optimisation studies which aim at rationalising crop acreage over a region to reach defined targets. Each crop or crop rotation is associated with a fixed sequence of technical operations which allows the calculation of various indicators such as input costs or externalities (de Juan et al., 1999; Nordblom et al., 2006; Reca et al., 2001; Stoorvogel, 1995).

However, Barson and Lesslie, (2004) and Leteinturier et al., (2006) note that detailing crop management systems and crop successions can be of importance in many environmental perspectives and Stomph et al. (1994) specify that a quantitative description of land use practices is necessary. This is why some studies, e.g. those reported in Table 1, use the concept of the cropping system to describe and locate agricultural activities over large areas.

# 3 Objectives of Describing and Locating Cropping Systems on a Regional Scale

This description of cropping systems may directly provide the solution to the management problem, or may only be a means to obtain this solution. For instance, to estimate the impact of land use on water resource use, the key is to know the distribution of irrigations during summer, i.e. to know the area cultivated with irrigated crops and the amount of water applied during the season (Leenhardt et al., 2004). However, most agri-environmental problems are relative to the impact of cropping systems on their environment. They deal with externalities of cropping systems, e.g. nitrate loss and pollen dispersal. Externalities are usually simulated by models using cropping system information as input. Depending on the physical processes involved, two modelling approaches exist. The first consists of spatially-explicit modelling to represent flows within the area under consideration (Dunning et al., 1995). The second consists of running a 1-D model such as

a crop model at every point, or a sample of points, of the area (Hartkamp et al., 2004). For spatially-explicit models the exact location of cropping systems is required, while for 1D models only the relative area grown with each crop  $\times$  management combination for each soil  $\times$  weather condition is necessary. Both modelling approaches require high-resolution land use data (Verburg et al., 2002). For instance, to estimate water pollution by nitrate leaching, it is necessary to model the nitrogen use by crops, nitrogen transformation in the soil, and vertical and lateral water flows (e.g. Beaujouan et al., 2001; Gomez and Ledoux, 2001). Such a modelling approach needs input data on the crops sown and their location, their sowing dates, that determine the periods of crop growth and water and nutrient uptake, and the periods of fertiliser and water supply. Describing and locating cropping systems in space thus meets a scientific need. It makes it possible to use on a regional scale models developed on a local scale (Faivre et al., 2004; Leenhardt et al., 2006).

We can distinguish three different types of issues where cropping system information is needed. First, diagnostic studies for evaluating the impact of land use and land cover on the environment or on natural resources require data on existing or past land use, and therefore, on existing or past cropping system distribution. Second, scenario simulation approaches also evaluate impacts, but those of proposed prospective rather than actual situations. Therefore, the required data not only concerns the current distribution of land use but also options and constraints for modifications in the current practices. Since prospective situations are usually compared with the present or a former situation, scenario studies are often associated with diagnostic studies. The last type aims at identifying optimal land use for a given economic, environmental or multicriteria objective. Optimal land use, or even optimal cropping system distribution, is determined by iterative choice and evaluation algorithms and two kinds of data are therefore required. First, a set of cropping systems is necessary to initiate the run. This set can correspond to the current situation, but this is not compulsory. Second, a set of alternative cropping systems, or a range of cropping system components to be combined during evaluation, must be specified. As in scenario simulation, the search for optimal solutions belongs to a planning approach and aims at determining the spatial organisation of cropping systems to reach desirable objectives.

To summarise, describing and locating cropping systems is used for diagnostic or scenario studies, either because cropping systems are directly targeted or because cropping system information is required as input for an environmental model. Describing and locating cropping systems in space allows multisimulations on a regional scale. Describing and locating cropping systems can also serve to define the base situation and cropping system basket from which an optimal land use can be calculated to satisfy given objective functions.

#### 4 Scale Issues

Up to this point the word "scale" has been used in its colloquial sense. It covers, in fact, both the "extent", i.e. the area of interest of the study, and the "support", comprising a finite number of smaller areas, the "support units", on which information is collected. Support units can cover integrally or not the extent; the ratio "support"/"extent" is called the coverage (Bierkens et al., 2000; Faivre et al., 2004).

The analysis of various case studies, and particularly those reported in Table 1, shows that the extent is most often determined by the processes involved, but also by the context and objectives of the study. For instance, when the studied environmental problem requires accounting for physical flows conditioned by topographical features, e.g. modelling water flows for calculating diffuse pollution, the extent corresponds to a space with natural landscape delimitations, e.g. a stream or a river catchment area (cases B & D, Table 1). In addition, the resulting pollution can be more easily measured at the outlet of the catchment, which also justifies choosing as the extent a whole catchment. In other situations, the process involved is not the only factor that determines the extent of the study. It is particularly the case when the study is conducted with a socio-economic partner who often influences the choice of the extent. For instance, in a study dealing with harvest purity (case A), the extent was partly determined by the expected impact of the studied process, here the area of pollen dissemination. Similarly, Gomez and Ledoux (2001) studied nitrate pollution for the whole Seine basin, rather than in a small catchment area, because it was the area of the water manager who was both partner and recipient of the study (case D). In the regional irrigation demand

study (case C), no physical dispersal or diffusion process was involved. Therefore, the extent was entirely determined by management considerations and corresponded to a space delimitated by installations, the irrigated area, which is the decision unit for water allocation decisions.

Once the extent is determined, it must be broken down into support units, where data on the cropping system is provided. The choice of support units often depends on the process involved and the model used to simulate it.

When dispersal and other spatial processes have to be considered, their nature and impact distance are often the key factor for choosing the support units. For example, when output flows from the cropping systems convey water and other elements that may influence surrounding crops (e.g. gene flow, herbicide runoff), it is important to simulate these lateral flows with spatially-explicit models, and to account for the spatial arrangement of crops and cropping systems. The field, which can be considered as a homogeneous unit regarding the crop and its management, naturally becomes the support unit for providing data on the cropping system, as well as the simulation unit (case A) Some models require as a simulation unit a sub-catchment  $(1-5 \text{ km}^2)$ , which then becomes the support unit of cropping system information. However, when the extent is very large relative to the size of fields or small sub-catchments, spatial processes may be negligible compared with other processes. For instance, Ledoux et al. (2007) neglected lateral flows between fields because they assumed that these lateral flows did not change the overall diffuse pollution. Neglecting spatial processes led to using a 1-D model to simulate "vertical" nitrate leaching. The only constraint for running these 1-D models on the studied area is to define spatial simulation units that are homogeneous combinations of the variables used as model input (generally variables describing the soil, the weather and the cropping system). These homogeneous soil-climate-cropping system combinations should naturally become the support units on which information regarding the cropping system has to be provided. This is, though, not always possible when extents are very large (case D) because the number of simulation units may be excessive regarding the computation time (Ledoux et al., 2007). Support units were then chosen a priori because they retrieved correctly the spatial differentiations of cropping systems (Mignolet et al., 2004), and the 1-D model was run on

homogeneous simulation units that were then not explicitly located within each support unit. Sometimes, no process model is required and conditions the choice of support units. For example, in case C, the objective of the study is to estimate the regional irrigation demand. This target can be reached by simply aggregating the irrigation demand estimated for each support unit, whatever its size and nature. Thus, in this case, the support unit can be determined a posteriori, with regard to the kinds of indicators used to estimate the spatial distribution of agricultural practices (Maton et al., 2007a), or a priori, with regard to their significance for the various stakeholders concerned with the study (Clavel and Leenhardt, 2008).

To conclude, we note that when the objective of the studies is to estimate externalities of cropping systems a process model is required. Then the process determines the extent and the support unit. When socioeconomic considerations prevail, as well as when no model is involved, it is often considerations related to stakeholders which condition extent and support unit choices.

# 5 Describing Cropping Systems by Relevant Variables

Describing a cropping system is a difficult task, even on the field scale. Should all technical operations of the crop management system be characterised? How detailed should this description be? This problem is even greater on a regional scale. Can cropping systems from two different sites be considered as similar? The elements that constitute a cropping system are the crop rotation and the various technical operations of the management system of each crop of the rotation. Which then are the elements to survey in order to describe the main representative cropping systems of a region?

The necessary cropping system variables must therefore be carefully chosen in relation to the study objective and the analysed processes to optimise the compromise between key variables and data availability. For instance, Colbach (2008), working on spatio-temporal gene flow in oilseed crops and volunteers (case A), showed that it is sufficient to include those cropping system components that significantly influence the two major stages for gene flow, i.e. plant survival and pollen dispersal, and that the remaining techniques such as fungicides or fertiliser, which mostly influence seed production, can be neglected. Similarly, in a study dealing with regional irrigation demand in a region dominated by irrigated maize (case C), Maton (2006) simplified the various cropping systems in irrigated maize to the 3 main elements influencing the total irrigation applied on a maize field, i.e. irrigation management, sowing practice and earliness choice. Such approaches reduce the amount of data to be collected to characterise the cropping systems without neglecting their internal consistency. They also help to simplify the simulation models used in the studies, rendering them thus more robust.

## 6 Collecting Data

Data collection for characterising and locating cropping systems on a regional scale is a real challenge. As a consequence, exhaustive and systematic collection of information about land management practices on the regional scale is only at its early stage, and is most of the time partial and descriptive, with no understanding of the links between technical interventions within crop management systems (BRS, 2006; Mignolet et al., 2007).

Two ways of collecting data to describe cropping systems are possible and not exclusive: the direct way, which aims at collecting information regarding the relevant elements of the cropping systems, and the indirect way, which consists of collecting information about indicators, or determining factors, of the cropping systems. These two ways, and the techniques used, are more or less adequate, depending on whether the cropping system data collection aims at describing existing or past cropping system distributions for elaborating a diagnosis, at building scenarios of cropping systems, or at seeking an optimal cropping system distribution.

# 6.1 Direct Collection

Collecting directly the variables that characterise the cropping systems concerns mainly diagnosis studies. Direct data collection by survey requires long and arduous work because of the great number of farms
on a regional scale and is thus generally unrealistic (Biarnès et al., 2004). Three main alternative solutions exist. The first is to use remote-sensing techniques to estimate the spatial variability of specific technical interventions, such as sowing dates (Launay and Guérif, 2005). However, some interventions and technical choices are not detectable by remote-sensing, or with difficulty or at high cost. This is the case, for example, for irrigation application dates and amounts or for the choice of cultivar or earliness for a given crop. The second is to make the best use of existing regional databases (e.g. Agricultural census, Ter-Uti-LUCAS database) to provide knowledge on past cropping systems, mostly by using statistical analysis. For example, Mari and Le Ber (2005) developed data-mining techniques to estimate the spatial variability of past crop rotations from such databases (Le Ber et al., 2006). This solution is, however, limited by the nature of variables collected systematically in administrative databases, and by the length and cost of specific surveys that may be conducted to get variables not systematically surveyed by administrations. If, in France, the number of variables collected by the many surveys conducted by the Department of Economical and Statistical Studies (SCEES<sup>2</sup>) is huge, the nature and size of the sample used differ from one survey to another, rendering multivariate analyses difficult if not impossible. The last option is to use expert knowledge, as Mignolet et al. (2004) did, to associate a management system with a crop or a crop rotation (case D). However, this option requires long surveys when the extent is large and if the spatial variability is to be taken into account, and the quality of expertise decreases as the survey goes further into the past. Experts can, however, inform not only on past and/or recommended practices, but also provide innovative practices that can be of interest for scenario studies (Lançon et al., 2008).

## 6.2 Indirect Collection: Use of Determining Factors

Despite the reduced number of variables to be collected for describing a cropping system, these variables may not be included in available databases, or not at a sufficient resolution for characterising past cropping systems on all support units. An indirect way to determine and locate the cropping systems within the extent is to identify the factors involved in their spatial organisation, e.g. soil depth and water availability for choosing crop species or soil type for choosing tillage strategies. These "determining factors" of the cropping system are often easier to collect on the support units and are thus used as indicators of current or past cropping systems for diagnosis studies. For scenario studies, the use of determining factors seems compulsory since it is not possible to collect cropping system data for future or hypothetical context changes. Knowing which factors determine the cropping system and estimating the evolution of such factors through modelling or expertise contributes to estimating and proposing cropping systems for future or hypothetical conditions. For instance, when cropping systems are determined by farming system or farm structure (case A - Angevin et al., 2002 - and C - Maton et al., 2007b), it is possible to deduce the distribution of cropping systems within a region from economic scenario analyses that produce changes in farm-type distributions (Zimmermann et al., 2006). Similarly, when cropping system elements can be related to climatic indicators (e.g. irrigation amounts and dates - Leenhardt et al., 2004, or sowing dates – Maton et al., 2007a), the use of climate change scenarios (e.g. long-term emission scenarios developed by the Intergovernmental Panel on Climate Change - IPCC, 2007) can provide indications on the way cropping systems over a region may change. Determining factors are also used in optimisation studies: they represent the constraints that should be satisfied when calculating the optimal crop distribution for a given objective function. For example, Rounsevell et al. (2003) consider that various variables determine (or constrain) the choice of cropping systems: the availability of machinery and labour on the farm, the existence of a sugar factory and regional sugar beet quotas, the previous crops, the soil and the climate, the latter two variables conditioning the workable hours.

Two kinds of "determining factors" may be used: external or internal factors. "External" factors include the characteristics of the physical environment, e.g. soil, topography or weather conditions; of the fields, e.g. shape and size; of the farm, e.g. farming systems, equipment and manpower; or of the socio-economic and administrative environment:

<sup>&</sup>lt;sup>2</sup> Service Central d'Études Économiques et Statistiques

professional networks, extension services and municipality. While maps, remote-sensing shots, land registers or interpolation procedures provide georeferencing for most factors related to the physical environment or field characteristics (Faivre et al., 2004), the last two groups of factors, those related to farms and the socio-economic environment, are available in various administrative databases which provide a location indication, most often the municipality the farm belongs to. For instance, Biarnès and Colin (2006) used the municipality to predict practices over the whole study extent since they found, by analysing field data samples, that weed control practices were well correlated with this factor (case B). In case A (Colbach, 2008; Angevin et al., 2002) the type and location of crop successions was considered as determined by the type of farm, defined by intensive vs. organic management, and by the farm and field sizes. Grid weather data were used by Maton et al.(2007a) to predict the spatial variability of sowing dates and by Leenhardt et al. (2004) to predict the spatial variability of irrigation dates (case C). Using an "external" determining factor consists of using a simple "If-then" allocation rule:

#### If [external indicator(s)]

#### then [cropping system (or element of it)]

However, sometimes no deterministic relation can be identified between a set of potential explanatory variables and the cropping systems or elements of them. This can be illustrated by Maton et al. (2007b), and Biarnès and Colin (2006). Maton et al. investigated determining factors of maize sowing and cultivar choice from geo-referenced databases (geographical information systems containing environmental characteristics of fields, spatially interpolated weather data series, or administrative databases containing farm characteristics collected from censuses). However, the variability of cropping systems could not be completely explained by these geo-referenced indicators. A combination of indicators led only to a probability of occurrence of a given cropping system. Similarly, (Biarnès and Colin, 2006) linked each municipality of the study extent to a given distribution of weed control practices.

The rule (1) becomes then a stochastic allocation rule:

If [external indicator(s)]

then [probabilities of cropping systems (or elements of them)]

An "internal" factor corresponds to an element of the cropping system on which the choice of the other elements of this cropping system depends. For instance, Maton et al. (2007b) showed that the choice of cultivar earliness was partly determined by the sowing date, while Colbach (2008) and Angevin et al. (2002) used the crop succession to determine crop management choices. Internal factors are, by definition, not geo-referenced, and must be indirectly related to external indicators in order to predict the cropping system location. A second-order rule is thus needed:

If [external indicator(s)]

then [element(s) E of the cropping system]

And

If [element(s) E of the cropping system]

then [cropping system]

The use of determining factors is therefore a way to overcome the difficulty of accessing data on cropping system variables. Identifying variables that can be used as determining factors of cropping systems can be done either by expertise from extension services or farm surveys, or by statistical analysis of databases. However, to specify cropping systems on all support units of the study extent, it is also necessary to be able to determine the modalities of these determining factor variables for all support units. Although determining factors are chosen among variables more easily accessible than cropping system variables, this can be problematic. For instance, in case A, cropping systems were recognised as determined by farm types, but the fields (support units) were not associated with a farm type. The solution was then to allocate randomly farm types to fields, while respecting certain spatial constraints such as isolation distances between crops.

To conclude, identifying determining factors of cropping systems can simplify the collection of cropping system information, but it is not always possible to obtain deterministic relations between easily accessible factors and elements of the cropping systems. In some cases, the access to the modalities of determining factors for all support units can also be difficult.

#### 7 Conclusion

Sustainable development of regions and sustainable management of resources often require actions concerning land use. Sustainable solutions need to account for the coherence of technical choices within cropping systems and with their environment. However, very few agro-environmental studies account for both the coherence and the spatial variability of cropping systems. It was the objective of this article to review methods to describe and locate cropping systems within large and/or heterogeneous areas. This article focused particularly on 4 case studies differing in their objectives, but also their extent and support. Describing and locating cropping systems is required for diagnostic or scenario needs, where environmental models are often used to simulate cropping system impact. We noticed that extents and support units were constrained by the objective of the study and, if a model was required, by the process to be modelled. Choosing the extent, and the nature and size of support units is crucial because it conditions not only the precision of the location of cropping systems but also the precision of their description. Indeed, on a regional scale, it is not possible to represent all aspects of a cropping system. A simplification is necessary. The variables strictly necessary to describe the cropping systems must be chosen in accordance with the study objective and with the analysed processes in order to optimise the compromise between key variables and data availability. Collecting information to fill all support units with the modalities of such variables may be done using administrative surveys, remote sensing or expertise. However, it appears that, very often, such sources of information are not sufficient to fill all support units or to get all necessary cropping system variables. The use of determining factors, that are easily geo-referenced, is a way to overcome the difficulty of accessing data on cropping systems. However, it is not always possible to obtain deterministic relations between easily accessible factors and elements of the cropping systems, and sometimes, the access to the modalities of determining factors for all support units can also be difficult. A common feature of the methods presented here is their reliance on expertise or on detailed survey data. Expertise regarding agricultural practices is available in most countries from extension services. However, its reliability decreases when the size of the support unit or the time period increase. Furthermore, detailed

surveys that are necessary either to describe cropping systems directly or to relate agricultural practices to their potential determining factors are rare and difficult to carry out for a single study. This calls for the development of adequate and routine land management practice monitoring.

Acknowledgements The authors are grateful to Drs. V. Souchère, C. Thenail and B. Nicoullaud, and to the anonymous reviewers for their valuable comments, recommendations and corrections. We would like to thank M. R. Lesslie and D. Wallach for editorial advice in English. This study was conducted within the framework of the Cropping System network of the Environment and Agronomy Department of the French National Institute for Agronomical research (INRA).

#### References

- Angevin F., Colbach N., Meynard J.M., Roturier C. (2002) Analysis of necessary adjustements of farming practices, in: Bock A.K., Lheureux K., Libeau-Dulos M., Nilsagard H., Rodriguez-Cerezo E. (Eds.), Scenarios for co-existence of genetically modified, conventional and organic crops in European agriculture. Technical Report Series of the Joint Research Center of the European Commission, EUR 20394 EN, http://ipts.jrc.ec.europa.eu/publications/pub.cfm? id=1044.
- Angevin F., Klein E.K., Choimet C., Gauffreteau A., Lavigne C., Messean A., Meynard J.M. (2008) Modelling the impacts of cropping systems and climate on maize cross-pollination in agricultural landscapes: The MAPOD model, Eur. J. Agron. 28, 471–484.
- Barson M., Lesslie R. (2004) Land management practices why they are important and how we know this, in: National workshop on Land Management Practices: Information Priorities, Classification and Mapping, Canberra, pp. 1–9.
- Beaujouan V., Durand P., Ruiz. L. (2001) Modelling the effect of the spatial distribution of agricultural practices on nitrogen fluxes in rural catchments, Ecol. Model. 137, 93–105.
- Benoît M., Papy F. (1998) La place de l'agronomie dans la problématique environnementale, Dossier de l'environnement de l'INRA 17, 53–62.
- Biarnès A., Colin F. (2006) Methodology for assessing the hydrological impact of weed control practices with a view to management of Mediterranean vine growing catchments, Int. J. Sust. Dev. World 9, 161–179.
- Biarnès A., Rio P., Hocheux A. (2004) Analyzing the determinants of spatial distribution of weed control practices in a Languedoc vineyard catchment, Agronomie 24, 187–196.
- Bierkens M.F.P., Finke P.A., de Willigen P. (2000) Upscaling and downscaling methods for environmental research, Kluwer Academic Publishers.
- BRS (2006) Guidelines for land use mapping in Australia: principles, procedures and definitions. A technical handbook supporting the Australian Collaborative Land Use Mapping Programme, Edition 3, Australian Government, Bureau of Rural Sciences, Canberra.

- Chipanshi A.C., Ripley E.A., Lawford R.G. (1999) Large-scale simulation of wheat yields in a semi-arid environment using a crop-growth model, Agr. Syst. 59, 57–66.
- Clavel L., Leenhardt D. (2008) Development of matrices to build scenarios of cropping systems distribution for integrated catchment assessment. A proposition for an irrigated area in south-western France, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A., Guariso G. (Eds.), Integrating Sciences and Information Technology for Environmental Assessment and Decision Making, iEMSs 2008: International Congress on Environmental Modelling and Software, 4th Biennial Meeting of iEMSs, pp. 678–685.
- Colbach N. (2008) How to model and simulate the effects of cropping systems on population dynamics and gene flow at the landscape level, Example of oilseed rape volunteers and their role for co-existence of GM and non-GM crops, submitted to Environ. Sci. Pollut. R.
- Donet I. (1999) Étude sur l'interpolation de l'ETP quotidienne en points de grille. Report SCEM/SERV/AGRO, Météo-France, Toulouse.
- Dunning J.B., Stewart D.J., Danielson B.J., Noon B.N., Root T.L., Lamberson R.H., Stevens E.E. (1995) Spatially explicit population models: current form and future uses, Ecol. Appl. 5, 3–11
- Faivre R., Bastié C., Husson, A. (2000) Integration of VEGETA-TION and HRVIR into yield estimation approach, in: Gilbert Saint (Ed.), Proceedings of "Vegetation 2000, 2 years of operation to prepare the future", Space Application Institute and Joint Reasearch Center, Ispra, Lake Maggiore (Italy), pp. 235–240.
- Faivre R., Leenhardt D., Voltz M., Benoît M., Papy F., Dedieu G., Wallach D. (2004) Spatialising crop models, Agronomie 24, 205–217.
- Gomez E., Ledoux E. (2001) Démarche de modélisation de la dynamique de l'azote dans les sols et de son transfert vers les aquifères et les eaux de surface, C.R. Acad. Agric. Fr. 87, 111–120.
- Hartkamp A.D., White J.W., Rossing W.A.H., van Ittersum M.K., Bakker E.J., Rabbinge R. (2004) Regional application of a cropping systems simulation model: crop residue retention in maize production systems of Jalisco, Mexico, Agr. Syst. 82, 117–138.
- Heinemann A.B., Hoogenboom G., Faria de R.T. (2002) Determination of spatial water requirements at county and regional levels using crop models and GIS . An example for the State of Parana, Brazil, Agr. Water Manage. 52, 177–196.
- IPCC (2007) Climate Change 2007: Mitigation of Climate Change. Contribution of Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Metz B., Davidson O.R., Bosch P.R., Dave R., Meyer L.A. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 851 p.
- Jouve P. (2006) La dimension spatiale des systèmes de culture: comparaison entre agriculture tempérée et agriculture tropicale, Cah. Agric. 15, 255–260.
- de Juan J.A., Tarjuelo J.M., Ortega J.F., Valiente M., Carrion P. (1999) Management of water consumption in agriculture: A model for the economic optimisation of water use: application to a sub-humid area, Agr. Water Manage. 40, 303–313.

- Lal H., Hoogenboom G., Calixte J.P., Jones J.W., Beinroth F.H. (1993) Using crop simulation models and GIS for regional productivity analysis, T. ASAE 36, 175–184.
- Lançon J., Reau R., Cariolle M., Munier-Jolain N., Omon B., Petit M.S., Viaux P., Wery J. (2008) Élaboration à dire d'experts de systèmes de culture innovants, in: Reau R., Doré T. (Eds.), Systèmes de culture innovants et durables : quelles méthodes pour les mettre au point et les évaluer ? Educagri éd., Dijon, pp. 91–107.
- Launay M., Guérif M. (2005) Assimilating remote sensing data into a crop model to improve predictive performance for spatial applications, Agr. Ecosyst. Environ. 321–339.
- Le Bail M. (2005) Le bassin d'approvisionnement : territoire de la gestion agronomique de la qualité des productions végétales, in: Prevost P. (Ed.), Agronomes et territoires : Deuxième édition des entretiens du Pradel, L'Harmattan, Paris, pp. 213–228.
- Le Ber F., Benoît M., Schott C., Mari J.F., Mignolet C. (2006) Studying crop sequences with CARROTAGE, a HMM-based data mining software, Ecol. Model. 191, 170–185.
- Ledoux E., Gomez E., Monget J.M., Viavattene C., Viennot P., Ducharne A., Benoit M., Mignolet C., Schott C., Mary B. (2007) Agriculture and groundwater nitrate contamination in the Seine basin. The STICS-MODCOU modelling chain, Sci. Total Environ. 375, 33–47.
- Leenhardt D., Trouvat J.L., Gonzalès G., Pérarnaud V., Prats S., Bergez J.E. (2004) Estimating irrigation demand for water management on a regional scale. I. ADEAU-MIS, a simulation platform based on biodecisionnal modelling and spatial information, Agr. Water Manage. 68, 207–232.
- Leenhardt D., Wallach D., Le Moigne P., Guérif M., Bruand A., Casterad M.A. (2006) Using crop models for multiple fields, in: Wallach D., Jones J., Makowski D. (Eds.), Working with crop models, Elsevier, pp. 209–248
- Leteinturier B., Herman J.L., Longueville F.d., Quintin L., Oger R. (2006) Adaptation of a crop sequence indicator based on a land parcel management system, Agr. Ecosyst. Environ. 112, 324–334.
- Louchart X., Voltz M., Andrieux P., Moussa R. (2001) Herbicide Transport to Surface Waters at Field and Watershed Scales in a Mediterranean Vineyard Area, J. Environ. Qual. 30, 982–991.
- Mari J.F., Le Ber F. (2005) Temporal and spatial data mining with second-order hidden markov models, Soft Computing – A Fusion of Foundations, Methodologies and Applications, 406–414.
- Martin P., Joannon A., Mignolet C., Souchère V., Thenail C. (2006) Systèmes de culture et territoires : cas des questions environnementales, in: Doré T., Le Bail M., Martin P., Ney B., Roger-Estrade J. (Eds.), L'agronomie aujourd'hui. Coll. Quae, Versailles.
- Maton L. (2006) Représentation et simulation des pratiques culturales des agriculteurs à l'échelle régionale pour estimer la demande en eau d'irrigation. Application à un bassin versant maïsicole du sud-ouest de la France, PhD thesis, INPT, Toulouse.
- Maton L., Bergez J.E., Leenhardt D. (2007a) Modelling the days which are agronomically suitable for sowing, Eur. J. Agron. 27, 123–129.

- Maton L., Leenhardt D., Bergez J.E. (2007b) Geo-referenced indicators of maize sowing and cultivar choice for better water management, Agron. Sustain. Dev. 27, 377–386.
- Meynard J.M., Doré T., Habib R. (2001) L'évaluation et la conception de systèmes de culture pour une agriculture durable, C.R. Acad. Agric. Fr. 87, 223–236.
- Mignolet C., Bornerand C., Benoît M. (2001) Dynamique spatiale et temporelle de l'activité agricole dans le bassin de la Seine au cours des trente dernières années, C.R. Acad. Agric. Fr. 87, 99–109.
- Mignolet C., Schott C., Benoît M. (2004) Spatial dynamics of agricultural practices on a basin territory: a retrospective study to implement models simulating nitrate flow. The case of the Seine basin, Agronomie 24, 219–236.
- Mignolet C., Schott C., Benoit M. (2007) Spatial dynamics of farming practices in the Seine basin: Methods for agronomic approaches on a regional scale, Sci. Total Environ. 375, 13–32.
- Moen T.N., Kaiser H.M., Riha S.J. (1994) Regional yield estimation using a crop simulation model: concepts, methods and validation, Agr. Syst. 46, 79–92.
- Nordblom T., Hume I., Bathgate A., Reynolds M. (2006) Mathematical optimisation of drainage and economic land use for target water and salt yields, Aust. J. Agr. Resour. Ec. 50, 381–402.
- Reca J., Roldan J., Alcaide M., Lopez R., Camacho E. (2001) Optimisation model for water allocation in deficit irrigation systems: I. Description of the model, Agr. Water Manage. 48, 103–116.
- Rosenthal W.D., Hammer G.L., Butler D. (1998) Predicting regional grain sorghum production in Australia using spatial data and crop simulation modelling, Agr. Forest. Meteorol. 91, 263–274.
- Rounsevell M.D.A., Annetts J.E., Audsley E., Mayr T., Reginster I. (2003) Modelling the spatial distribution of

agricultural land use at the regional scale, Agr. Ecosyst. Environ. 95, 465–479.

- Sebillotte M. (1974) Agronomie et agriculture. Essai d'analyse des tâches de l'agronome, Cah. ORSTOM 24, 3–25.
- Sousa V. Santos Pereira L. (1999) Regional analysis of irrigation water requirements using kriging. Application to potato crop (Solanum tuberosum L.) at Tras-os-Montes, Agr. Water Manage. 40, 221–233.
- Stomph T.J., Fresco L.O., Van Keulen H. (1994) Land use system evaluation: Concepts and methodology, Agr. Syst. 44, 243–255.
- Stoorvogel J.J. (1995) Integration of computer-based models and tools to evaluate alternative land-use scenarios as part of an agricultural system analysis, Agr. Syst. 49, 353–367.
- Veldkamp A., Kok K., de Koning G.H.J., Schoorl J.M., Sonneveld M.P.W., Verburg P.H. (2001) Multi-scale system approaches in agronomic research at the landscape level, Soil Till. Res. 58, 129–140.
- Verburg P.H., Soepboer W., Veldkamp A., Limpiada R., Espaldon V., Mastura S.S.A. (2002) Modeling the Spatial Dynamics of Regional Land Use: The CLUE-S Model, Environ. Manage. 30, 391–405.
- Verburg P.H., Veldkamp A. (2001) The role of spatially explicit models in land-use change research: a case study for cropping patterns in China, Agr. Ecosyst. Environ. 85, 177–190.
- Yun J.I. (2003) Predicting regional rice production in South Korea using spatial data and crop-growth modelling, Agr. Syst. 77, 23–38.
- Zimmermann A, Heckelei T., Pérez I. (2006) Working paper – Literature Review of Approaches to Estimate Structural Change, SEAMLESS Report No.16, SEAMLESS integrated project, EU 6th Framework Programme, contract No. 010036-2, http://www.SEAMLESS-IP.org, 45 p.

# Part II Food Security

## **Nutritional Quality and Safety of Organic Food**

Denis Lairon

## Contents

1	Introduction				
2	Nutritional Quality of Organic Foodstuffs		101		
	2.1	Dry Matter Content	101		
	2.2	Macronutrient Contents	101		
	2.3	Mineral Contents	101		
	2.4	Vitamin Contents	102		
	2.5	Other Phytomicronutrients	102		
3	Sanitary Assessment of Organic Foodstuffs				
	3.1	Pathogenic Microorganisms	103		
	3.2	Phytochemical Contaminants	104		
	3.3	Mycotoxins	104		
	3.4	Nitrates	105		
4	Conc	lusion	106		
References					

D. Lairon (🖂)

Abstract Food security, nutritional quality and safety vary widely around the world. Reaching these three goals is one of the major challenges for the near future. Up to now, industrialized production methods have clearly shown severe limitations such as a worldwide contamination of the food chain and water by persistent pesticide residues, and reduced nutrient and flavor contents through low-cost intensive food production and/or processing. In line with several published literature reviews, the French Agency for Food Safety (AF-SSA) performed under my coordination an up-to-date exhaustive and critical evaluation of the nutritional and sanitary quality of organic food. This review is based on the AFSSA report issued and recently published studies. The major points are: 1/ organic plant products contain more dry matter and minerals (Fe, Mg); and contain more anti-oxidant micronutrients such as phenols and salicylic acid, 2/ organic animal products contain more polyunsaturated fatty acids, 3/ data on carbohydrate, protein and vitamin levels are insufficiently documented, 4/94-100% of organic food does not contain any pesticide residues, 5/ organic vegetables contain far less nitrates, about 50% less; and 6/ organic cereals contain overall similar levels of mycotoxins as conventional ones. Thus, organic agricultural systems have already proved able to produce food with high quality standards. I propose also improvements of organic production to achieve sustainable food production for humans in the near future.

**Keywords** Sustainable agriculture • Organic agriculture • Human food • Nutrition • Food safety • Contaminants • Health

INRA, UMR 1260, Nutriments Lipidiques et Prévention des Maladies Métaboliques, U476, Univ. Aix-Marseille 1, Univ. Aix-Marseille 2, Faculté de Médecine, 13385 Marseille, France e-mail: denis.lairon@univmed.fr

## 1 Introduction

While mass foodstuff production was achieved in industrialized countries during the twentieth century, the limitations of such an intensive production system have been highlighted for decades by ecologists and numerous agronomists, nutritionists and medical doctors. Briefly, great concern has been caused by high energy and chemical inputs, worldwide contamination of the food chain and water by persistent pesticide residues and nitrates, and the reduced nutrient and flavor contents through low-cost and intensive food production and extensive milling or processing. Only recently has the combined awareness for environment protection, food safety and security and well-being markedly raised public concern and demand for ecologically grown staple foods (El-Hage Scialabba, 2007; Niggli et al., 2007). For developing countries, concern is also focused on the appropriate way to ensure present and future food security, the number of malnourished and undernourished people approaching one billion worldwide, with no decreasing trend for the coming decade (FAO, http://www.fao.org). Worldwide, emphasis is increasingly being put on the relationship between food, nutrition and health (WHO, 2004; WCRF, 2007).

In fact, the nutritional and toxicological value of food produced under methods of ecological agriculture has long been a matter of interest and debate. Despite the potential importance of this topic for human wellbeing, only a limited number of studies have been specifically carried out due to the past general lack of consideration of alternative and sustainable methods of food production. During the last few decades, several literature reviews have already been performed and published in this field (Schuphan, 1974; Finesilver et al., 1989; Lairon et al., 1984a; Woëse et al., 1997; Worthington, 1998; Food Standards Agency, 2000; Soil Association, 2001; Brandt and Mölgaard, 2001; Bourn and Prescott, 2002). Our AFSSA report was issued in 2003 (AFSSA, 2003) and some new reviews have recently been published (Magkos et al., 2006; Winter and Davis, 2006; Rembialkowska, 2007). In most cases, these reviews have used data from original studies or previous reviews without true consideration of the quality of the data. The conclusions derived can somewhat differ but they generally highlight some benefits from organic agriculture.

In 2001, the French Agency for Food Safety (Agence française de sécurité sanitaire des aliments (AFSSA), http://www.afssa.fr) aimed to perform an exhaustive and critical evaluation of the nutritional and sanitary quality of organic food. To this aim, an expert working group was set up under my coordination. We aimed to reach the highest quality standard during our evaluation. We thus defined inclusion as well as exclusion criteria for use of original publications. Briefly, selected papers should refer to welldefined and certified organic agricultural practices, and have necessary information on design and follow-up, valid measured parameters and appropriate sampling and statistical analyses. After more than two years of work involving about 50 experts from all specific areas including organic agriculture, a final consensus report was issued in the French language in 2003 (AFSSA, 2003). The present review paper is a summary of this report including some updating and some more personal suggestions.

In all circumstances, organic agriculture is first defined as by the European Union regulation (CCE/2092/91 and CE/1804/99) and secondly by the French regulation for animal productions. To summarize, the main characteristics of the organic agriculture production system are respect for the environment and animals, promotion of sustainable cropping methods, use of non-chemical fertilizers and pest/disease/weed control means, production of high-quality foodstuffs and no use of genetically modified (GM) crops.

The limit of such an evaluation is the insufficient number of studies published in this area. For some aspects, the available studies allow one to reveal some trend or conclusion. In some others, too limited information of sufficient quality hinders any sound assessment.

In the first part of this review, the nutritional value of organic food will be described comparatively with that of conventional food. This includes the dry matter contents of fruit and vegetables, macronutrients, minerals and vitamins in various staple foods, and phyto-microconstituants, especially antioxidants. In the second part, sanitary properties of organic foodstuffs will be reported. Contaminations by pathogenic microorganisms, phytochemical contaminants or mycotoxins, and nitrate levels are reported. In the Conclusion, the main data obtained are discussed in the context of sustainable agriculture development, with some specific suggestions for further improving food quality.

## 2 Nutritional Quality of Organic Foodstuffs

#### 2.1 Dry Matter Content

The available data mostly refer to vegetables and fruit. For leafy vegetables as well as root vegetables and tubers, a trend for higher dry matter contents in organic foodstuffs has been found while no significant difference has been identified for fruit vegetables and fruit (Woëse et al., 1997; Bourn and Prescott, 2002; AFSSA, 2003).

## 2.2 Macronutrient Contents

Data regarding the levels of starch and carbohydrates in organic food are too limited to allow any conclusion.

For protein levels, it has been reported that organically-grown cereals, especially wheat, can have comparable protein levels with conventional ones (Shier et al., 1984) but generally have somewhat lower levels of protein than the conventional ones (Woëse et al., 1997). Nonetheless, it is noteworthy that the cultivars selected by organic farmers are mostly high-protein ones (e.g. for bread-making) and that optimized fertilization practices can maintain reasonably high protein levels. Moreover, a 25–30% increase in lysine has been reported in organic wheat (Wolfson and Shearer, 1981; Brandt et al., 2000). Comparative studies performed on hen eggs (Kouba et al., 2002) and raw cow's milk (Toledo et al., 2002) did not show any noticeable difference in protein levels.

For lipids, few studies have compared the total lipid content of beef, pork or chicken meat (Pastsshenko et al., 2000; Hansson et al., 2000; Hönikel, 1998; Fischer, 2001; Castellini et al., 2002). It is acknowledged that meat from cows and sheep contains less fat when animals are fed with grass rather than concentrate. In that line, a study conducted in Sweden showed that organically-bred cows have more lean meat than their conventional counterparts (Hansson et al., 2000). This was not found in pigs (Sundrum et al., 2000). More qualitatively, feeding cows with grass compared with concentrate led to a four-fold higher muscle content of linolenic acid, a recommended essential fatty acid of the n-3 series, with a concomitant decrease in oleic and linoleic acids (Nuernberg et al., 2002). Meat from organicallygrown cows has more polyunsaturated fatty acids (Pastsshenko et al., 2000). Castellini et al. (2002) showed that chickens of the same strain raised under an organic husbandry system have meat containing twoto three-fold less abdominal fat with 2-3 times less fat in the filet and 1.8 times less fat in the leg. Moreover, the n-3 fatty acid content in the filet was significantly higher with no difference for saturated fatty acids.

The total milk fat was not overall different, while some studies highlighted the higher levels of polyunsaturated fatty acids (Lund, 1991). Clearly, cow diet is a determinant of tissue and milk fatty acid levels, grazing or ingesting silages modulating the levels of polyunsaturated fatty acids as well as trans-fatty acids and conjugated linoleic acid (Ferlay et al., 2006).

A single study has shown that organic virgin olive oil has a higher oleic acid level (Gutierrez et al., 1999).

## 2.3 Mineral Contents

The most important mineral elements are calcium (Ca), magnesium (Mg), potassium (K), iron (Fe), zinc (Zn), copper (Cu), manganese (Mn), selenium (Se) and iodine (I). Phosphorus (P) and sodium (Na) are generally found in sufficient quantity.

Fruit and vegetables. 22 scientific publications were considered in the AFSSA report (2003). Regarding fruit, and especially apples, it is noteworthy that the mineral composition is generally not noticeably altered by the production system. Regarding vegetables (potato, carrot, beetroot, lettuce, kale, leek, turnip, onion, celeriac and tomato), a trend has been observed for higher levels of iron and magnesium expressed on a fresh matter basis in organic foodstuffs, with no other marked change.

Cereals. From two long-term fertilization trials, it appears that the mineral composition (P, K, Ca, Mg, Mn, Zn, Fe, Cu and Cr) of cereals is not markedly affected

by the cropping regime (Miller and Dema, 1958; Morel et al., 1984). Another study did not show any marked difference but a trend for higher levels of Ca, Cu and Zn in organic barley (Alföldi et al., 1996).

In a recent review (Rembialkowska, 2007), it was estimated that organic crops overall contain 21% more iron and 29% more magnesium than their conventional counterparts.

Animal products. Very limited information is available. For milk, an evaluation of three different studies did not allow the identification of a difference due to the husbandry system. The same conclusion was reached by Woëse et al. (1997) from four comparative studies performed on meat. Nevertheless, it has been clearly shown that chickens grown in open fields compared with housing have somewhat higher iron levels (Castellini et al., 2002).

## 2.4 Vitamin Contents

The number of studies dedicated to vitamin contents is limited to some fruits and vegetables and eggs. Regarding water-soluble vitamins, the most studied one has been Vitamin C (ascorbic acid), a key vitamin for which higher daily intakes are recommended. Studies performed on potato (Fischer and Richter, 1986; Kolbe et al., 1995), tomato (Pither and Hall, 1990; Caris-Veyrat et al., 2004), celeriac (Leclerc et al., 1991) and kale showed higher vitamin C levels in organically-grown products. In contrast, no difference was found during studies in leek, carrot or beetroot. A study on apple did not show any difference either (Weibel et al., 2000). Very sparse and inconclusive data have been published on vitamin B1 and B2 levels.

Fat-soluble vitamin and carotenoid contents have been the subjects of some studies. A higher vitamin E level in organic olive oil has been found in one study (Gutierrez et al., 1999). A review by Woëse et al. (1997) related 27 studies reporting on  $\beta$ -carotene levels in vegetables and no noticeable differences were found overall between organic and conventional foodstuffs. Brandt and Mölgaard (2001) reported a positive relationship between N-fertilization and  $\beta$ -carotene levels in carrots, while a recent study on organic vs. conventional tomatoes showed higher contents of  $\beta$ carotene (Caris-Veyrat et al., 2004). Another report (Martin et al., 2002) has shown that a grass-rich regimen compared with a maize silage or concentrate generates milk with a higher vitamin E and  $\beta$ -carotene content.

#### 2.5 Other Phytomicronutrients

Fruit and vegetables contain a large variety of microcompounds which are secondary metabolites in plants such as polyphenols, resveratrol and some non-provitaminic carotenoids. These compounds have increasingly been shown to have drastic regulatory effects at cellular level and are thus involved in prevention of certain diseases such as cancers, chronic inflammation and other pathologies. Some of them are phytoalexins which are produced in plants as a response to external stress such as fungal disease. While several factors can modulate their plant level such as cultivar, maturity, light or temperature, some studies have compared the levels of some of these phytomicrocompounds in fruit or vegetables depending on the cropping system. For phenols and polyphenols, a majority of studies showed higher levels in organic foodstuffs such as apple (Lucarini et al., 1999), peach (Carbonaro et al., 2002), pear (Carbonaro et al., 2002), potatoes (Hamouz et al., 1999), onion (Ren et al., 2001), tomato (Mitchell et al., 2007), pepper (Pérez-López et al., 2007), orange (Tarozzi et al., 2006) and olive oil (Gutierrez et al., 1999), while some others did not show any difference. It has been estimated in a recent review (Rembialkowska, 2007) that organic plant foods overall contain double the amount of phenolic compounds. One study reported higher levels of resveratrol in organic wines (Levite et al., 2000).

The median contents of salicylic acid in organic vegetable soups were significantly higher (117 vs. 20) than in the compared non-organic ones (Baxter et al., 2001). Organically-grown tomatoes also have a higher salicylic acid content than conventional ones (Rossi et al., 2008). It is noteworthy that salicylic acid is the active anti-inflammatory compound of aspirin.

While some better anti-oxidant and antiproliferative positive effects on cancer cells have been observed with organic vs. conventional extracts (Tarozzi et al., 2006; Olsson et al., 2006) the effects of chronic diets in humans have yet to be fully investigated (Grinder-Pedersen et al., 2003).

Increased contents	Reduced contents	Comparable contents
Dry matter in vegetables	Pesticide residues in all food (mostly absent)	Mycotoxins in cereals & milk
Some minerals (iron, magnesium) in vegetables	Nitrates in vegetables	Most minerals in fruit, vegetables & cereals
Anti-oxidants in crops: Vitamin C (potatoes) Polyphenols in fruit & vegetables, Salicylic acid in vegetables		Beta-carotene in fruit & vegetables
Polyunsaturated fatty acids in meat and milk	Saturated fatty acids in meat	
Most nutrients in wholegrain organic cereals and derivatives	Protein content in grains	

 Table 1
 Key items of nutritional and sanitary value of organic compared with conventional food

Table 1 highlights the key items of nutritional and sanitary value of organic compared with conventional food.

## 3 Sanitary Assessment of Organic Foodstuffs

Risk assessment is a scientific approach aiming at identifying known hazards and related risks. Contaminations by bacteria, viruses, worms, mycotoxins and agro-chemicals are mainly involved. Not all aspects have yet been comparatively studied such as hazards due to viruses or worms. We will therefore take some relevant examples on other aspects based on more reliable information.

#### 3.1 Pathogenic Microorganisms

Plant products. Concern has been raised by the EU Scientific Committee on Food on the possible contamination of fruit and vegetables by sewage sludges, animal manures or irrigation waters (SCF, 2002). Indeed, they can provide pathogenic microorganisms such as bacteria pathogenic to humans (Strauch, 1991), *Listeria monocytogenes* (Van Renterghem et al., 1991) or *Salmonella sp.* (Warnick et al., 2001). In fact, sewage sludges are not allowed for use in organic agriculture and fresh manures are not used as such for fertilization but composted for variable periods of time. It has been shown that the aerobic composting process, including a high temperature phase, has the capability of markedly reducing or fully eliminating the pathogenic microorganisms initially present such as *Salmonella*  *Enteritidis* or *E. coli* (Lung et al., 2001; Droffner and Brinton, 1995; Vuorinen and Saharinen, 1997; Tiquia et al., 1998). This could be less efficient in eliminating *Clostridium botulinum* (Bohnel and Lube, 2000). In addition, when added to the soil, the exogenous bacteria are quickly eliminated due to unfavorable conditions (Van Renterghem et al., 1991; Dowe et al., 1997). In conclusion, it appears that the systematic use of aerobic composting is a suitable way to maximize the hygienic properties of organic fertilizers, and thus to avoid significant contamination of organic foodstuffs by pathogenic microorganisms.

Animal products. A survey conducted in Austria (Zangerl et al., 2000) evaluated the degree of bacterial contamination of dairy products using two indicators (E. coli and S. aureus) and did not find different levels in organic or conventional foodstuffs. Another survey conducted in France (Echevarria, 2001) in 1997-99 in four different regions also found comparable levels for total bacteria count or butyric microorganisms in milks produced with the two husbandry systems. In fact, the limited use of silage in organic husbandry could be beneficial for reducing the possible contamination of ruminant feed by Listeria monocytogene E. coli O157s (Herriott et al., 1998). In contrast, a Danish study reported that about 100% of poultry samples were contaminated by Campylobacter sp in organic farms, whereas 36-49% of samples in conventional farms were (Heuer et al., 2001).

To summarize, there is no convincing information indicating that organic foodstuffs can be differently contaminated than conventional ones. This is likely the result of efficient preventive measures. Enteric viruses from five families (*Picornaviridae*, *Caliciviridae*, *Astroviridae*, *Reoviridae* and *Adenoviridae*) are known to be harmful to humans but there is a lack of comparative studies on this aspect.

#### 3.2 Phytochemical Contaminants

The banned use of toxic chemical pesticides, fungicides and herbicides in organic agriculture systems is clearly a gold standard in terms of protection of land workers' health and environment biodiversity and well-being. Anyway, the question has repeatedly been raised of the level of contamination of organic foodstuffs by environmental pollution. Several surveys have been dedicated to this matter. The SETRABIO survey (SETRABIO, 2000) was conducted in France in 1993-99 on 15,772 samples of raw or processed foodstuffs, mostly cereals. 94% of organic samples were devoid of any contaminant residue while 3.3% only contained levels clearly above the detection level but far less than legal maximum levels. During the years, a trend has been observed for diminishing contamination levels (i.e. 1.7% in 1998-99 vs. 4.4% in 1993-97). The French DGAL/COOPAGRI/ESMISAB survey (2001) was then conducted in 1999-2001 on 1500 samples. No residue of 78 potential phytochemical contaminants was detected.

Another study performed on vegetables and strawberries in Sweden did not show any contamination of organic ones, while 17–50% of conventional ones contained residues (Bourn and Prescott, 2002). Results from the monitoring of pesticide residues in fruit and vegetables on the Danish market in 2000–01 found that only 2.8% of organic samples were contaminated by pesticide residues, all being below the MRL (Poulsen and Andersen, 2003). A recent survey conducted in Italy in the 2002–2005 period on 3500 samples of food of plant origin concluded that the vast majority (97.4%) of organic farming products do not contain detectable pesticide residues (Tasiopoulou et al., 2007).

In contrast, it is known that a large proportion of usual foodstuffs are contaminated by phytochemical residues. This has clearly been shown by the recent reports by the EU DG SANCO for 2001 and 2005. The 2005 report (DG SANCO, 2007) was based on 62 500 samples collected in EU member states and analyzed for 706 chemicals. 41% of samples were shown to be contaminated, with 4.7% of samples with levels above the legal maximum levels (MRL).

In fact, all these phytochemicals have been fully or permanently authorized for use and residue levels considered harmless have been set up on the basis of available toxicity data. Because these molecules have a high toxic capacity (including mutagenesis and carcinogenesis) and the long-term effects of chronic lowdose ingestion are not known in humans, the consumer and scientific concern about possible health damage is real. The EU REACH project is aimed at better evaluating their toxicity and deleting the most risky ones. In that context, organic products clearly show a real advantage.

It is worthwhile mentioning that some natural extracts are used in organic agriculture for pest and disease control such as pyrethrins, rotenone, copper salts and sulfur. The first two are quickly degraded in the fields and no contaminations of foodstuffs have been observed (Moore et al., 2000). Regarding copper and sulfur, their use is limited to surface spraying and copper amounts have been progressively reduced. Contamination levels have not been investigated.

## 3.3 Mycotoxins

Mycotoxins are a large family of toxic molecules synthetized by molds developing on plants such as *Aspergillus*, *Penicillium* and *Fusarium*. Most are highly toxic and heat-resistant and can be transferred along the alimentary chain from plants to animals then human food. The most recognized in terms of public health hazard are aflatoxins, Ochratoxin A (OTA), Fumonisins, Deoxynivalenol (DON), Patulin and Zearalenone. Overall, they display strong harmful effects such as immunotoxicity, teratogenesis, embryotoxicity, nephrotoxicity and liver cancerogenesis. They all have very low legal maximum levels in food (0.1–2 ppb/kg body weight).

Given the lack of use of chemicals (including chemical fungicides) in organic production, the question has been raised of a possible high level of contamination of organic productions. Several surveys have been conducted to test this hypothesis. A study in Germany was conducted on cereals in 1997–98 (Birzele et al., 2000), indicating that most organic samples were contaminated by DON and a minority by OTA, but that conventional samples were comparably contaminated. A French study searched for contamination of cereals by several mycotoxins and observed that conventional ones were frequently contaminated at low levels while organic ones were less frequently contaminated but at high levels in a few cases (Malmauret et al., 2002). Wheat contamination has also been tested in a controlled DOC trial in Switzerland with mean DON levels of 74 ppb in organic and 109 ppb in conventional wheat (Kuhn, 1999). A study in Denmark (Jörgensen et al., 1996) showed a trend for higher mean OTA levels in organic cereals but with the highest levels recorded in conventional ones. A recent study showed that organic wheat was less contaminated by *Fusarium* and contained less ZEN and DON than conventional wheat: when fed to pigs, the bile samples from organically-fed pigs contained lower concentrations of ZEN (Schneweis et al., 2005).

Some other studies have been conducted on wheat flour, providing comparable or lower levels of mycotoxin contamination for organic flour. Finally, processed cereal products such as bread, muesli and biscuits have been tested (Parent-Massin et al., 2002). Overall, almost half of the organic products were found to be contaminated to variable degrees but no comparison was made with conventional ones. When a comparison was made, lower levels of contamination were found in organic ones (Usleber et al., 2000). No marked differences in OTA levels were found in cereal derivatives in Italy (Biffi et al., 2004) or in cereal-based baby food (Beretta et al., 2002). Two studies have observed lower levels of aflatoxin in organic milks compared with conventional ones (Gravert et al., 1989; Frank Hansen, 1990), whereas another one found some high levels in organic milk (Ghidini et al., 2005). Organic beers collected in Belgium during 2003-2004 were more frequently OTA-contaminated than their conventional counterparts but this difference was not found in 2005 (Anselme et al., 2006). Contamination of beers by DON was marginal.

In conclusion, contamination of foodstuffs, especially cereals, is widespread but at a low level and an organic or conventional mode of production do not lead to overall noticeable differences. In fact, the preventive measures used in organic systems, despite the non-use of fungicides, appears generally able to maintain contamination at a low level.

#### 3.4 Nitrates

Nitrates are a matter of concern for public health due to their easy transformation into nitrites. Nitrites are highly reactive molecules capable of i/ competing with oxygen in blood circulation for binding to hemoglobin, thus leading to methemoglobinemia and possible anoxia and, ii/ binding to secondary amines to generate nitrosamines which are among the most powerful natural cancer-promoting moities. For that reason, maximum daily intakes for nitrates (3.7 mg/kg body weight) and nitrites (0.07 mg/kg body weight) have been set up by the FAO/OMS JECFA, along with a maximum nitrate level in drinking water (50 mg/L).

In the human diet, about 80% of nitrates are provided by vegetables, while nitrate levels in fruits, cereals and legumes are very low (French Inventaire National de la Qualité Alimentaire, 1982; Stopes et al., 1988; Cornée et al., 1992). Animal products contain very low levels of nitrates while processed meat can contain added nitrites as a preservative.

Nitrates are naturally present in plants; they are absorbed through the roots and further used for aminoacid synthesis. They can accumulate in plant tissues, especially in vegetables. Several comparative studies have been performed on nitrate levels in vegetables.

At the level of a retail shop, we performed a study on five vegetables in spring and observed significantly lower nitrate contents (-28 to -85%) in organic potato, leek, turnip and salad but not in organic kale (Lairon et al., 1982). A comparable study performed in Austria on 17 vegetables found lower nitrate contents (-40% to -86%) in organic ones except in spinach (Rauter and Wolkerstorfer, 1982). In Germany, a comparison on carrots showed 61% less nitrates in organic ones (Pommer and Lepschy, 1985).

At farm level, by comparing designated crops on matched farms, three studies provided interesting data. We performed one in Provence (Lairon et al., 1985) and found in the organically-grown samples –39% nitrates in lettuce, –46% in potato, –22% in carrot and a higher content in one sample for leek. In Switzerland, organic lettuces grown over two years contained 2.5 times less nitrates than their conventional counterparts in May–June, 1.2 times less in October and comparable high levels in November (Temperli et al., 1982; Vogtmann et al., 1984). In contrast, two other studies performed on tomato in Israel (Basker, 1992) and carrot in Norway (Hogstad et al., 1997) did not show noticeable differences.

Fertilization trials have also been designed to compare the effects of fertilization regimes on nitrate contents in vegetables. Overall, composts compared with chemical fertilizers lead to lower nitrate accumulations in most vegetables such as lettuce, potato, carrot, turnip, leek, beetroot and spinach (Lairon et al., 1984a, 1984b; Mäder et al., 1993) while an absence of difference can also be found on a few occasions (Vogtmann et al., 1984). Nitrogen-rich organic fertilizers can also generate lower nitrate contents, but when mineralization conditions are very favorable they can also lead to high nitrate accumulations (Lairon et al., 1985; Termine et al., 1987). Finally, it is noteworthy that some natural nitrogen-rich fertilizers such as Chili sodium nitrate can stimulate nitrate accumulation in sensitive vegetables (Lairon et al., 1984b). Indeed, it is clear that the level of nitrate in vegetables results from nitrogen availability for roots, temperature, light exposure, and cultivars and species. The use of organic fertilization with slowly or moderately available nitrogen (especially composts) is key to explaining the generally observed lower nitrate accumulation in organic vegetables.

From the above data, it appears that depending on seasons, organic vegetables can overall contain at least 30–50% less nitrates than conventional ones. Because the habitual average level of nitrate intake is in the range 120–280 mg/d (French Inventaire National de la Qualité Alimentaire, 1982; Cornée et al., 1992; Stopes et al., 1988) and close to the maximum daily dose defined by the FAO/OMS JECFA (i.e. 220 mg/60 kg person), and a marked increase in vegetable and fruit consumption is widely recommended (WHO, 2004; PNNS, 2001), organic vegetables can make this recommended increase safer regarding the nitrate issue.

#### 4 Conclusion

The present review based on available scientific literature highlights that organic plant products tend to have more dry matter, some minerals (Fe, Mg) and antioxidant micronutrients (phenols, resveratrol) while animal organic products have more polyunsaturated fatty acids. Regarding safety issues, the vast majority (94–100%) of organic food does not contain any pesticide residues, organic vegetables contain markedly less nitrates (about half) and organic cereals contain overall comparable levels of mycotoxins with conventional ones. The conclusions of this new review are in line with those of most reviews published on this matter.

This critical literature review indicates that organic agriculture, as developed until now, has the potential to produce high-quality products with some relevant improvements in terms of contents of anti-oxidant phytomicronutrients, nitrate accumulation in vegetables and toxic phytochemical residue levels.

After decades of smooth increase in organic production, a sharp rise in consumer demand and producer awareness is occurring now. The nutritional and toxicological data I reported here should encourage such a new trend. Indeed, I think that organic agricultural systems, recently recognized as highly efficient and sustainable ones (El-Hage Scialabba, 2007), are now facing the challenge of turning from a "niche" into a potential long-term worldwide impact. This implies a global development strategy for a sustainable organic food chain as summarized in Table 2. To successfully

Agricultural management	Food production	Food access and security	Food quality	Consumer attitude
Soil fertility management Preventive crop	Reliable certification	Fair trade	High sensory value High putrient	Confidence
protection	Sustained optimal	Local food access	content	awareness and demand
Animal welfare	yields		Controlled pathogens	
Environment and biodiversity protection	Low inputs	Optimized supply chain	No pesticide residues	Health protection
Farmer	Sufficient	Affordable		Sustained
well-being	income	retail price		well-being

 Table 2 Key aspects of a sustainable organic food chain

achieve this goal new and significant support for the development of these sustainable systems should be raised. For instance, raising new cultivars and crops suited to low-input systems and more resistant to diseases, along with optimized nutrition value, is one of the biggest challenges for future sustainable agriculture worldwide. This may be performed in the context of sustained biodiversity rather than GM productions, which are not allowed in organic/ecological farming systems. Regarding nutritional aspects, I suggest that several research lines are developed in the near future to improve sustainable food production systems such as increasing the levels of dry matter in foodstuffs to optimize nutrient density and intakes, increasing the levels of recommended n-3 fatty acids and lower saturated ones, increasing levels of limiting minerals, vitamins and anti-oxidants, improving the taste and flavor of fruit and vegetables to stimulate awareness, developing the use of wholemeal/partlyrefined cereal flours rich in fibers and nutrients and the sourdough fermentation process for optimized nutritional efficiency, further limiting accumulation of nitrates in vegetables and mycotoxins in cereal products, and improving producers', consumers' and stakeholders' knowledge in the food production-diet-health chain.

## References

- AFSSA (2003) Report on Evaluation of the nutritional and sanitary quality of organic foods (Evaluation nutritionnelle et sanitaire des aliments issus de l'agriculture biologique, in French), AFSSA, 164 p. Available on line at http://www.afssa.fr.
- Alföldi T., Mader P., Niggli U., Spiess E., Dubois D., Besson J.-M. (1996) Quality investigation in the long term DOC trial, Quality of plant products growth with manure fertilisation, Proceeding of the Fourth meeting (Juva Finland, 6–9 July 1996 Darmstadt Germany), Institut for Biodynamic research, pp. 34–43.
- Anselme M., Tangni E.K., Pussemier L., Motte J.C., Van Hove F., Schneider Y.J., Van Peteghem C., Larondelle Y. (2006) Comparison of ochratoxin A and deoxynivalenol in organically and conventionally produced beers sold on the Belgian market, Food Add. Contam. 23, 910–918.
- Basker D. (1992) Comparison of taste quality between organically and conventionally grown fruits and vegetables, Am. J. Alternative Agric. 7, 129–136.
- Baxter G.J., Graham A.B., Lawrence J.R., Wiles D., Paterson J.R. (2001) Salicylic acid in soups prepared from organically and non-organically grown vegetables, Eur. J. Nutr. 40, 289–292.

- Biffi R., Munari M., Dioguardi L., Ballabio C., Cattaneo A., Galli C.L., Restani P. (2004) Ochratoxin A in conventional and organic cereal derivatives: a survey of the italian market 2001–02, Food Addit. Contam. 21, 586–591.
- Birzele B., Prange A., Kramer J. (2000) Deoxynivalenol and ochratoxin A in german wheat and changes of level in relation to storage parameters, Food Addit. Contam. 17, 1027–1035.
- Bohnel H., Lube K. (2000) Clostridium botulinum and biocompost. A contribution to the analysis of potential health hazards caused by bio-waste recycling, J. Vet. Med. B 47, 785–795.
- Bourn D., Prescott J. (2002) A comparison of the nutritional value, sensory qualities and food safety of organically and conventionally produced foods, Crit. Rev. Food Sci. Nutr. 42, 1–34.
- Brandt K., Mölgaard J.P. (2001) Organic agriculture: does it enhance or reduce the nutritional value of plant foods? J. Sci. Food Agr. 81, 924–931.
- Brandt D.A., Brand T.S., Cruywagen C.W. (2000) The use of crude protein content to predict concentrations of lysine and methionine in grain harvested from selected cultivars of wheat, barley and triticale grown in the western cape region of South Africa, S. Afr. J. Anim. Sci. 30, 22–25.
- Carbonaro M., Matterra M., Nicoli S., Bergamo P., Cappelloni M. (2002) Modulation of antioxydant compounds in organic vs. conventional fruit (peach, Prunus persica L., and pear, Pyrus communis L.), J. Agr. Food Chem. 50, 5458–5462.
- Caris-Veyrat C., Amiot M.J., Tyssandier V., Grasselly D., Buret M., Mikolajczak M., Guilland J.C., Bouteloup-Demange C., Borel P. (2004) Influence of organic versus conventional agricultural practice on the antioxidant microconstituent content of tomatoes and derived purees; consequences on antioxidant plasma status in humans, J Agr. Food Chem. 52, 6503–6509.
- Castellini C., Mugnai C., Dal Bosco A. (2002) Effect of organic production system on broiler carcass and meat quality, Meat Sci. 60, 219–225.
- Cornée J., Lairon D., Velema J., Guyader M., Berthezene P. (1992) An estimate of nitrate, nitrite and N- nitrosodimethylamine concnetrations in french food products or food groups, Sci. Alim. 12, 155–197.
- DGAL/COOPAGRI/ESMISAB (2001) Évaluation de l'exposition des consommateurs et produits issus de l'agriculture biologique et de l'agriculture conventionnelle aux résidus de pesticides, métaux lourds, nitrates, nitrites et mycotoxines, Notre Alimentation 37, I-VI.
- DG SANCO (2007) Monitoring of pesticide residues in products of plant origin in European union, Norway, Iceland and Liechtenstein, 2005 Report. http://ec.europa.eu/food/ fvo/specialreports/pesticide\_residues/report\_2005\_en.pdf.
- Dowe M.J., Jackson E.D., Mori J.G., Bell C.R. (1997) Listeria monocytogenes survival in soil and incidence in agricultural soils, J. Food Prot. 60, 1201–1207.

- Droffner M.L., Brinton W.F. (1995) Survival of E. coli and Salmonella populations in aerobic thermophilic composts as measured with DNA gene probes, Zbl. Hyg. 197, 387–397.
- Echevarria L. (2001) Qualité du lait livré par les élevages agrobiologiques de quatre régions françaises, Renc. Rech. Rum. 8, 95.
- El-Hage Scialabba N. (2007) Organic agriculture and food security, Food and Agriculture Organization of the United Nations, Report of the International conference on organic agriculture and food security, May 3–5, Roma, Italy, 22 p. (available at http://www.fao.org).

FAO, http://www.fao.org.

- Ferlay A., Martin B., Pradel P., Coulon J.B., Chilliard Y. (2006) Influence of grass-based diets on milk fatty acid composition and milk lipolytic system in Tarentaise and Montbeliarde cow breeds, J. Dairy Sci. 89, 4026–4041.
- Finesilver T., Johns T., Hill S.B. (1989) Comparison of food quality of organically versus conventionally grown plant foods, a review, Ecological Agriculture Projects, Report, McGill University, Canada.
- Fischer J. (2001) Blé bio : De bonnes variétés boulangères pour la panification française, Perspect. Agric. n° 272.
- Fischer A., Richter C. (1986) Influence of organic and mineral fertilizers on yield and quality of potatoes, in: Vogtmann H., Boehnke E., Fricke I. (Eds.), The importance of biological agriculture in a world of diminishing resources (Proceedings of the 5th IFOAM Conference), Verlagsgruppe, witzenhausen, Germany, pp. 236–248.
- Food Standards Agency (2000) Food Standards Agency (UK) view on organic foods, 4 p.
- Frank Hansen L. (1990) Characterization of organic milk, Proceedings of the Ecological Agriculture NJF-seminar 166 – Miljövard.
- French Inventaire National de la Qualité Alimentaire (1982) Ministère de l'Environnement, Neuilly, France, 199 p.
- Ghidini S., Zanardi E., Battaglia A., Varisco G., Ferretti E., Campanini G., Chizzolini R. (2005) Comparison of contaminant and residue levels in organic and conventional milk and meat products from Northern Italy, Food Add. Contam. 22, 9–14.
- Gravert H.O., Pabst K., Ordolff D., Treitel U. (1989) Milcherzeugung in alternativen Landbau, K. Milchw. Forsch. 41, 211–223.
- Grinder-Pedersen L., Rasmussen S.E., Bügel S., Jørgensen L.V., Dragsted L.O., Gundersen V., Sandström B. (2003) Effect of diets based on foods from conventional versus organic production on intake and excretion of flavonoids and markers of antioxidative defense in humans, J. Agr. Food Chem. 51, 5671–5676.
- Gutierrez F., Arnaud T., Albi M.A. (1999) Influence of ecological cultivation on virgin olive oil quality, JAOCS 76, 617–621.
- Hamouz K., Lachman J., Vokal B., Pivec V. (1999) Influence of environmental conditions and way of cultivation on the polyphenol and ascorbic acid content in potato tubers, Rostlinna Vyroba 45, 293–298.
- Hansson I., Hamilton C., Ekman T., Forslund K. (2000) Carcass quality in certified organic production compared with conventional livestock production, J. Vet. Med. 47, 111–120.
- Herriott D.E., Hancock D.D., Ebel E.D., Carpenter L.V., Rice D.H., Besser T.E. (1998) Association of herd management

factors with colonization of dairy cattle by shiga toxinpositive Escherichia coli 0157, J. Food Protect. 61, 802–807.

- Heuer O.E., Perdersen K., Andersen J.S., Madsen M. (2001) Prevalence and antimicrobial susceptibility of thermophilic Campylobacter in organic and conventional broiler flocks, Lett. Appl. Microbiol. 33, 269–274.
- Hogstad S., Risvik E., Steinsholt K. (1997) Sensory quality and chemical composition of carrots: a multivariate study, Acta Agr. Scand. 47, 253–264.
- Hönikel K.O. (1998) Qualität ökologisch erzeugter Lebensmittel tierischer Herkunft., Dtsch. Tierärzt. Wschr. 105, 327–329.
- Jörgensen K., Rasmussen G., Thorup I. (1996) Ochratoxin A in Danish cereals 1986–92 and daily uptake by Danish population, Food Addit. Contam. 13, 95–104.
- Kolbe H., Meineke S., Zhang W.L. (1995) Institute for Plant Nutrition, Germany: Differences in organic and mineral fertilisation on potato tuber yield and chemical composition compared to model calculations, Agribiol. Res. 48, 63–73.
- Kouba M., Enser M., Whittington F.M., Nute G.R., Wood J.D. (2002) Effet d'un regime riche en acide linolénique sur les activités d'enzymes lipogéniques, la composition en acides gras et la qualité de la viande chez le porc en croissance, Neuvièmes Journées des Sciences du Muscle et Technologie de la Viande, 15–16 octobre 2002 - Clermont-Ferrand. http://www.ofival.fr/vpc/9jsmtv/9som.htm.
- Kuhn F. (1999) Bestimmung von Trichothecenen in Weizen aus verschiedenen Anbausystemen mittels HPLC-MS, Diplomarbeit, Universität Basel, 60 p.
- Lairon D., Lafont H., Léonardi J., Hauton J.C., Ribaud P. (1982) Comparaison de l'intérêt nutritif de légumes produits par l'agriculture conventionnelle ou biologique, Sci. Aliments 2(HS II), 203–205.
- Lairon D., Termine E., Lafont H. (1984a) Valeur nutritionnelle comparée des légumes obtenus par les méthodes de l'agriculture biologique ou de l'agriculture conventionnelle, Cah. Nutr. Diet. 6, 331–339.
- Lairon D., Spitz N., Termine E., Ribaud P., Lafont H., Hauton J.C. (1984b) Effect of organic and mineral nitrogen fertilization on yield and nutritive value of butterhead lettuce, Plant Foods Hum. Nutr. 34, 97–108.
- Lairon D., Termine E., Gauthier S., Lafont H. (1985) Teneurs en nitrates des productions maraîchères obtenues par des méthodes de l'agriculture biologique, Sci. Aliments 5(HS V), 337–343.
- Leclerc J., Miller M.L., Joliet E., Rocquelin G. (1991) Vitamin and mineral contents of carrot and celeriac grown under mineral or organic fertilization, Biol. Agric. Hortic. 7, 339–348.
- Levite D., Adrian M., Tamm L. (2000) Preliminary results of resveratrol in wine of organic and conventional vineyards, Proceedings of the 6th International Congress on organic Viticulture, Basel (Suisse), pp. 256–257.
- Lockeretz W. (1983) Environmentally sound agriculture, Proceedings of the 4th IFOAM International Conference Cambridge, USA, August 18–20, 1982, Praeger, 426 p.
- Lucarini M., Carbonaro M., Nicoli S., Aguzzi A., Cappelloni M., Ruggeri S., Di Lullo G., Gambelli L., Carnovale E. (1999) Endogenous markers for organic versus conventional plant products, Agri-Food Quality II: Quality Management of Fruits and Vegetables, pp. 306–310.
- Lund P. (1991) Characterization of alternatively produced milk, Milchwissenschaft 46, 166–169.

- Lung A.J., Lin C.M., Kim J.M., Marshall M.R., Nordstedt R., Thompson N.P., Wie C.I. (2001) Destruction of Escherichia Coli O157:H7 and Salmonella Enteritidis in cow manure composting, J. Food Protect. 64, 1309–1314.
- Mäder L., Pfiffner L., Niggli U., Balzer U., Balzer F., Plochberger A., Velimirov A., Boltzmann L., Besson J.M. (1993) Effect of three farming systems (bio-dynamic, bioorganic, conventional) on yield and quality of beetroot (Beta vulgaris L. var. esculenta L.) in a seven year crop rotation, Acta Hort. 339, 11–31.
- Magkos F., Arvaniti F., Zampelas A. (2006) Organic food: buying more safety or just peace of mind? A critical review of the litterature, Crit. Rev. Food Sci. Nutr. 46, 23–56.
- Malmauret L., Parent-Massin D., Hardy J.L., Verger P. (2002) Contaminants in organic and conventional foodstuffs in France, Food Addit. Contam. 19, 524–532.
- Martin B., Ferlay A., Pradel P., Rock E., Grolier P., Dupont D., Gruffat D., Besle J.M., Ballot N., Chilliard Y., Coulon J.B. (2002) Variabilité de la teneur des laits en constituants d'intérêt nutritionnel selon la nature des fourrages consommés par les vaches laitières, Actes des 9<sup>es</sup> Renc. Rech. Rum. 9, 347–350.
- Miller D.S., Dema I.S. (1958) Nutritive value of wheat from the Rothamsted Broadbalk field, Proc. Nutr. Soc. 17, xliv-xlv.
- Mitchell A.E., Hong Y.J., Koh E., Barrett D.M., Bryant D.E., Denison R.F., Kaffka S. (2007) Ten-year comparison of the influence of organic and conventional crop management practices on the content of flavonoids in tomatoes, J. Agr. Food Chem. 55, 6154–9.
- Moore V.K., Zabik M.E., Zabick M.J. (2000) Evaluation of conventional and "organic" baby food brands for eight organochlorine and five botanical pesticides, Food Chem. 71, 443–447.
- Morel R., Lasnier T., Bourgeois S. (1984) Les essais de fertilisation de longue durée de la station agronomique de Grignon, Ed. Inra, Paris, 335 p.
- Niggli U., Leifert C., Aföldi C., Lück L., Willer H. (2007) Proceedings of the 3rd International Congress of the European integrated project Quality Low Imput Food (QLIF), University of Hohenheim, March 20–23, 2007, http://www.qlif.org.
- Nuernberg K., Nuernberg G., Ender K., Lorenz S., Winkler K., Rickert R., Steinhart H. (2002) N-3 fatty acid and conjugated linoleic acids of longissimus muscle in beef cattle, Eur. J. Lip. Sci. Technol. 104, 463–471.
- Olsson M.E., Andersson C.S., Oredsson S., Berglund R.H., Gustavsson K.E. (2006) Antioxidant levels and inhibition of cancer cell proliferation in vitro by extracts from organically and conventionally cultivated strawberries, J. Agr. Food Chem. 54, 1248–1255.
- Parent-Massin D., Conan J.C., Langler C., Leveque J.M., Thisse M. (2002) Analysis of mycotoxin levels in bread, biscuits, muesli, apple juice and apple marmelade prepared from organic raw materials, Toxicol. Lett. 135, 107.
- Pastsshenko V., Matthes H.D., Hein T., Holzer Z. (2000) Impact of cattle grazing on meat fatty acid composition in relation to human nutrition, in Proceedings 13th IFOAM Scientific Conference, pp. 293–296.
- Pérez-López A.J., López-Nicolas J.M., Núñez-Delicado E., Del Amor F.M., Carbonell-Barrachina A.A. (2007) Effects of agricultural practices on color, carotenoids composition, and

minerals contents of sweet peppers, cv. Almuden, J. Agr. Food Chem. 55, 8158–8164.

- Pither R., Hall M.N. (1990) Analytical survey of the nutritional composition of organically grown fruit and vegetables, Campden, June 1990.
- PNNS, Plan National Nutrition Santé (2001–2011) http://www. sante.gouv.fr.
- Pommer G., Lepschy J. (1985) Investigation of the contents of winter wheat and carrots from different sources of production and marketing, Bayer. Landwirtsch. Jahrb. 62, 549–563.
- Poulsen M.E., Andersen J.H. (2003). Results from the monitoring of pesticide residues in fruit and vegetables on the Danish market, 2000-01, Food Addit. Contam. 20, 742–757.
- Rauter W., Wolkerstorfer W. (1982) Nitrat in gemuse, Z. Lebensm. Unters. F. 175, 122–124.
- Rembialkowska E. (2007) Quality of plant products from organic agriculture, J. Sci. Food Agr. 87, 2757–2762.
- Ren H., Bao H., Endo H., Hayashi T. (2001) Antioxidative and antimicrobial activities and flavonoid contents of organically cultivated vegetables, Nippon Shokuhin Kagaku Kaishi 48, 246–252.
- Rossi F., Godani F., Bertuzzi T., Trevisan M., Ferrari F., Gatti S. (2008) Health-promoting substances and heavy metal content in tomatoes grown with different farming techniques, Eur. J. Nutr. 47, 266–272.
- SCF (2002) Risk profile on the microbiological contamination of fruits and vegetables eaten raw. Report on the Scientific Committee on Food, European Commission, 29 Avril 2002. http://europa.eu.int/comm/food/fs/sc/ scf/out125\_en.pdf.
- Schneweis I., Meyer K., Ritzmann M., Hoffmann P., Dempfle L., Bauer J. (2005) Influence of organically or conventionally produced wheat on health, performance and mycotoxin residues in tissues and bile of growing pigs, Arch. Anim. Nutr. 59, 155–163.
- Schuphan W. (1974) Nutritional Value of crops as influenced by organic and inorganic fertiliser treatments, Qual. Plant. Plant Food. Hum. Nutr. 23, 333–358.
- SETRABIO (2000) Etude des teneurs de résidus en pesticides dans les produits biologiques bruts et transformés, 1993–2000, 115 p.
- Shier N.W., Kelman J., Dunson J.W. (1984) A comparison of crude protein, moisture, ash and crop yield between organic and conventionally grown wheat, Nutr. Rep. Int. 30, 71–73.
- Soil Association (2001) Organic farming, food quality and human health. A review of the evidence, 87 p.
- Stopes C., Woodward L., Forde G., Vogtmann H. (1988) The nitrate content of vegetable and salad crops offered to the consumer as from organic or conventional production systems, Biol. Agric. Hortic. 5, 215–221.
- Strauch D. (1991) Survival of microorganisms and parasites in excreta, manure and sewage sludge, Rev. Sci. Tech. Off. Int. Epiz. 10, 816–846.
- Sundrum A., Bütfering L., Henning M., Hoppenbrock K.H. (2000) Effects of on-farm diets for organic pig production on performance and carcass quality, J. Anim. Sci. 78, 1199–1205.
- Tarozzi A., Hrelia S., Angeloni C., Morroni F., Biagi P., Guardigli M., Cantelli-Forti G., Hrelia P. (2006) Antioxidant effectiveness of organically and non-organically grown red oranges in cell culture systems, Eur. J. Nutr. 45, 152–158.

- Tasiopoulou S., Chiodini A.M., Vellere F., Visentin S. (2007) Results of the monitoring program of pesticide residues in organic food of plant origin in Lombardy (Italy), J. Environ. Sci. Health B 42, 835–841.
- Temperli A., Kunsch V., Scharer H., Konrad P., Suter H., Ott P., Eichenberger M., Schmid O. (1982) Einfluss zweier Anbauweisen auf den Nitratgehalt von Kopfsalat, Schweiz. Landw. Fo. 21, 167–196.
- Termine E., Lairon D., Taupier-Letage B., Gauthier S., Lafont R., Lafont H. (1987) Yield and contents in nitrates, minerals and ascorbic acid of leeks and turnips grown under mineral or organic nitrogen fertilization, Plant Food. Hum. Nutr. 37, 321–332.
- Tiquia S.M., Tam N.F.Y., Hodgkiss I.J. (1998) Salmonella elimination during composting of spent pig litter, Bioresource Technol. 63, 193–196.
- Toledo P., Andren A., Bjrk L. (2002) Composition of raw milk from sustainable production systems, Int. Dairy J. 12, 75–80.
- Usleber E., Lepschy J., Mätlbauer E. (2000) Deoxynivaleol in Mehlproben des Jahres 1999 aus dem Einzelhandel, Mycotoxin Res. 16A, 30–33.
- Van Renterghem B., Huysman F., Rygole R., Verstraete W. (1991) Detection and prevalence of Listeria monocytogenes in the agricultural ecosystem, J. Appl. Bacteriol. 71, 211–217.
- Vogtmann H., Temperli A.T., Künsch U., Eichenberger M., Ott P. (1984) Accumulation of nitrates in leafy vegetables grown under contrasting agricultural systems, Biol. Agric. Hortic. 2, 51–68.
- Vuorinen A.H., Saharinen M.H. (1997) Evolution of microbiological and chemical parameters during manure and straw co-composting in a drum composting system, Agr. Ecosyst. Environ. 66, 19–29.

- Warnick L.D., Crofton L.M., Pelzer K.D., Hawkins M.J. (2001) Risk factors for clinical salmonellosis in Virginia, USA cattle herds, Prev. Vet. Med. 49, 259–275.
- Weibel F.P., Bickel R., Leuthold S., Alföldi T. (2000) Are organically grown apples tastier and healthier? A comparative field study using conventional and alternative methods to measure fruit quality, Acta Hort. 517, 417–426.
- WHO (2004) Food and health in Europe: a new basis for action. WHO Regional Publications, European series, No. 96, 385 p.
- Winter C.K., Davis S.F. (2006) Organic Foods, J. Food Sci. 71, R117-R124.
- Woëse K., Lange D., Boess C., Bögl K.W. (1997) A comparison of organically and conventionally grown foods - Results of a review of the relevant literature, J. Sci. Food Agr. 74, 281–293.
- Wolfson J.L., Shearer G. (1981) Amino acid composition of grain of maize grown with and without pesticides and standard commercial fertilizers, Agron. J. 73, 611–613.
- World Cancer Research Fund (WCRF) (2007) Food, nutrition, physical activity, and the prevention of cancer: a global perspective, American Institute of Cancer, Washington DC, USA, 517 p.
- Worthington V. (1998) Effect of agricultural methods on nutritional quality: a comparison of organic with conventional crops, Altern. Ther. 4, 58–69.
- Zangerl P., Ginzinger W., Tschager E., Lobitzer I. (2000) Sensory quality and microbial load of milk products from organic farming in Austria, in Proceedings 13th International IFOAM Scientific Conference, Basel, Switzerland, 28–31 August 2000, p. 298.

## Minerals in Plant Food: Effect of Agricultural Practices and Role in Human Health

M.C. Martínez-Ballesta, R. Dominguez-Perles, D.A. Moreno, B. Muries, C. Alcaraz-López, E. Bastías, C. García-Viguera, and M. Carvajal

## Contents

111
112
112
113
113
113
114
114
114
115
115
115
115
116
116
116
116
117
119
120
120
121
121
121
122
122
122
122
123
124
124

M. Carvajal (🖂)

Plant Nutrition Department, CEBAS-CSIC, PO Box 164, Espinardo, 30100 Murcia, Spain e-mail: mcarvaja@cebas.csic.es Abstract Interest in nutrient absorption and accumulation is derived from the need to increase crop productivity by better nutrition and also to improve the nutritional quality of plants as foods and feeds. This review focuses on contrasting data on the importance for human health of food mineral nutrients (Ca, Mg, K, Na and P) and also the trace elements considered essential or beneficial for human health (Cr. Co. Cu. Fe, Mn, Mo, Ni, Se and Zn). In addition, environmental stresses such as salinity, drought, extreme temperatures and light conditions that affect mineral content were revised in the light that the effect of these factors depends on the species or cultivar, and the specific plant organ, as well as the intensity and duration of the stress. Differences between inorganic and organic fertilisation practices on the mineral levels were also analysed to evaluate the influence of external factors on the quality of plant-based foods.

**Keywords** Environmental stress • Human health • Mineral fertilisation • Mineral nutrition

## **1** Introduction

One of the most important challenges for agriculture, besides enhancing food production, is to provide almost all the essential minerals and organic nutrients to humans for maintenance of health and proper organ function. Humans need more than 22 mineral elements; some of them are required in large amounts, but others, such as Fe, Zn, Cu, I and Se, are required in trace amounts because higher concentrations can be harmful (Welch and Graham et al., 2004; Grusak

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_8,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Martínez-Ballesta et al., Agron. Sustain. Dev. 30 (2010) 295–309. DOI: 10.1051/agro:2009022

M.C. Martínez-Ballesta et al.

and Cakmak, 2005). Although vegetables constitute the main source of minerals in the human diet, crops do not always contain sufficient amounts of these essential nutrients to meet dietary requirements (Welch et al., 1997). Elements that might enhance growth or that have a function in some plants (not in all plants) are referred to as beneficial elements. Concerning mineral nutrients, deficiencies, including those of Ca, Zn, Se, Fe and I, are almost certainly impairing the health and productivity of a large number of people in the developing world, especially poor women, infants and children (Graham et al., 2001). However, an excessive intake of minerals may also have a deleterious effect on the systemic physiology, that has led researchers in the last few years to acquire accurate data on the minimum requirements and toxic dosages of the minerals present in food. The level of minerals in vegetables depends on a number of factors including genetic properties of the crop species, climatic conditions, soil characteristics and the degree of maturity of the plant at the moment of harvesting.



**Fig. 1** Agricultural practices and environmental stresses affecting mineral composition in vegetables and fruits

Consideration of the environmental consequences and soil fertility practices are an essential component of the research in plant nutrition. Thus, some plant nutrients, such as potassium and sodium, are involved in plant responses to salt and water stress. Also, recommendations for amounts and application of fertilisers are continually modified to optimise the quality of the food production (Fig. 1).

## 2 Minerals in Foods of Plant Origin: Their Role in Human Health

Before attempting to modify the nutritional components in plants destined for human foods, careful consideration must be made in selection of minerals, their efficacy, and whether low or high dietary intake could have unintended negative health consequences. For selected mineral targets, the clinical and epidemiological evidence clearly plays a significant role in maintenance of optimal health, and they are limited in the diet worldwide (Lachance, 1998). In the following, we summarise the effect that essential or beneficial mineral nutrients have on human health.

## 2.1 Calcium

The concentration of calcium (Ca) in foods of plant origin shows a wide range of variation. The lower values belong to apples (Malus domestica), green pepper (Capsicum annuum) and potatoes (Solanum tuberosum) (<8.7 mg/100 g) and higher values are present in broccoli (Brassica oleracea L. var.italica) (100 mg/100 g) and spinach (Spinacia oleracea) (600 mg/100 g). Data on the mineral content of foods are important and should be considered when recommending the daily intake of minerals, as the Recommended Daily Allowance (RDA) for these nutrients is set out in the wide range of 800-1300 mg/day (http://www.anyvitamins.com/rda.htm). Calcium is an essential mineral for human health, participating in the biological functions of several tissues (musculoskeletal, nervous and cardiac system, bones and teeth, and parathyroid gland). In addition, Ca may act as a cofactor in enzyme reactions (fatty acid oxidation, mitochondrial carrier for ATP, etc.) and it is involved in the maintenance of the mineral homeostasis and physiological performance in general (Theobald, 2005; Huskisson et al., 2007; Morgan, 2008; Williams, 2008). Recent reports showed the unequivocal role of Ca as a second messenger (Morgan, 2008). With respect to disease prevention, Ca intake moderately reduces the risk of colon cancer (Pele et al., 2007; Peters et al., 2004). An increase in Ca intake during pregnancy is recommended to prevent risk of preeclampsia (Peters et al., 2004). Several studies have shown an association between suboptimal Ca intake and osteoporosis, hypercholesterolemia and high blood pressure (Unal et al., 2007). Although Ca levels undergo homeostatic controls to avoid an excessive accumulation in blood or tissues, there are a number of conditions that result in an excess of Ca within the body because of a failure in the control mechanisms: hypercalcaemia may occur as a result of either increased mobilisation of Ca from bone, or increased tubular reabsorption or decreased glomerular filtration in the kidneys, and less frequently, as the result of an increase in the dietary intake (Theobald, 2005).

### 2.2 Magnesium

Magnesium (Mg) has a strong presence in vegetable foods and also shows a critical role in the maintenance of human health through the diet. Vegetables and fruits contain, in general, Mg<sup>2+</sup> in the range of 5.5-191 mg/100 g fresh weight; and the recommended daily intake is 200-400 mg (http://www.anyvitamins. com/rda.htm). This essential mineral acts as a Ca antagonist on vascular smooth muscle tone and on postreceptor insulin signalling. It has also been related to energy metabolism, release of neurotransmitters and endothelial cell functions (Bo and Pisu, 2008). In addition, Mg participates with muscle and nerve excitability, as a cofactor of up to 300 enzymes (Huskisson et al., 2007). Magnesium deficiency is related to ageing and age-related disorders, mainly as a consequence of deficient intake in the diet (Durlach et al., 1998; Killilea and Maier, 2008). Recent findings showed that an increase in the intake of this mineral helps to protect people from the incidence of chronic diseases such as diabetes, metabolic syndrome, hypertension and several cardiovascular conditions (Bo and Pisu, 2008), where a low-Mg diet may contribute to insulin resistance, especially when this deficiency is

combined with a high-fructose diet. Moreover, reduced Mg intake is linked to inflammatory response as a result of modulation of the intracellular-Ca concentration (Ahokas et al., 2005; Rayssiguier et al., 2006). Magnesium toxic effects are not frequent, the most common side effects of an excessive intake of this mineral being headache, nausea, hypotension and unspecific bone and abdominal pain (Guerrero-Romero and Rodríguez-Morán, 2005).

## 2.3 Potassium

Foods of plant origin have potassium (K) contents of 20 to 730 mg/100 g fresh weight, although some plants such as 'Idaho' potatoes (S. tuberosum), banana (Musa spp.) and avocado (Persea americana) may all present high K contents (>700 mg/100 g fresh weight). Seeds and nuts are rich in K, showing values significantly higher than those mentioned above, up to 2240 mg/100 g. The recommended intake for this mineral is 3500 mg per day (http://www.anyvitamins.com/ rda.htm). Potassium plays a role in the maintenance of the balance of the physical fluid system and assisting nerve functions through its role in the transmittance of nerve impulses. It is also related to heart activity muscle contraction (Rosenthal and Gilly, 2003; Schwarz and Bauer, 2004; Ko et al., 2008; Lambert et al., 2008; Sobotka et al., 2008). However, K requirements are also dependent on the physiological or pathological moment. A deficiency may result in fatigue, cramping legs, muscle weakness, slow reflexes, acne, dry skin, mood changes and irregular heartbeat. Moreover, a reduced level of K produces alkalosis, which makes the kidney less able to retain this mineral. Excessive K can be toxic systemically when associated with hyperkaelemia in a catabolic state accompanied by oliguria (secondary to kidney failure) (Sobotka et al., 2008).

## 2.4 Sodium

Raw vegetables and fruit juices contain relatively low levels of sodium (Na) in the range of 2.28 to 94.0 mg/100 g and from 0.04 to 277 mg/100 g, respectively (Szefer and Grembecka, 2007). The role of Na in human physiology is related to the maintenance of the balance of physiological fluids (blood pressure, kidney function, nerve and muscle functions) (Sobotka et al., 2008; Hall, 2003; Hall et al., 1999; French and Zamponi, 2005). The recommended daily intake for Na is 2400 mg (http://www.anyvitamins.com/rda.htm). A deficiency is rare, but it can happen in cases of diarrhoea, vomiting or excessive sweating, and a shortage may lead to nausea, dizziness, poor concentration and muscle weakness, etc. (Smith et al., 2000; Soupart and Decaux, 1996). Excessive Na may be due to an increase in absorption or a secondary condition to kidney alteration, causing high blood pressure and neurological complications (Hall, 2003; Hall et al., 1999; Agrawal et al., 2008; Kahn, 2008). Excessive long-term use of Na may also cause a secondary loss of Ca.

## 2.5 Phosphorus

Phosphorus (P) is present in vegetables in the range of 16.2-437 mg/100 g. The lowest content of P is shown in fruits, which are in the range 9.9-94.3 mg/100 g (Szefer and Grembecka, 2007). The phosphorus daily recommended intake is 800-1300 mg (http://www. anyvitamins.com/rda.htm). Phosphate  $(PO_4^{3-})$  is required to produce ATP, GTP and CP as energetic substances and to regulate the activity of a number of proteins by means of phosphorylation reactions (Sobotka et al., 2008). Phosphorus is closely related to Ca homeostasis and also related to bone and teeth formation and the majority of the metabolic actions in the body, including kidney functioning, cell growth and the contraction of the heart muscle (Theobald, 2005; Szefer and Grembecka, 2007; Renkema et al., 2008). Deficiency of this element is unusual but symptoms are described as painful bones, irregular breathing, fatigue, anxiety, numbness, skin sensitivity and changes in body weight. If Ca supply is also deficient, then the condition may become severe because of increased risks of high blood pressure and bowel cancer. Ingesting dosages of P exceeding 3-4 g/day may be harmful as it can interfere with Ca absorption (Ghosh and Joshi, 2008; Moe, 2008).

## 2.6 Chromium

The concentration of chromium (Cr) generally ranges from  $4 \times 10^{-5}$  to  $6 \times 10^{-3}$  mg/100 g in vegetables

and 0.005 to 0.018 mg/100 g in fruits (Szefer and Grembecka, 2007). A RDA for Cr is not well defined, but it is considered to be between  $25-35 \mu g/day$ , fruits and vegetables being the major dietary contributors of Cr intake (http://www.anyvitamins.com/ rda.htm). Because of its 'micronutrient' characteristics, it is difficult to differentiate its content in foods from improper food contaminations (Lukaski, 2004). It is well accepted that Cr is essential for normal blood glucose and lipid metabolism and an insulincoadjuvant (Huskisson et al., 2007; Lukaski, 2004; Shenkin, 2008). Other biochemical actions for Cr such as involvement in gene expression, energy production, lipoprotein or lipid synthesis and metabolism regulation have been also described (Shenkin, 2008). Deficiencies in Cr are accompanied by glucose intolerance, weight loss and peripheral neuropathy (Shenkin, 2008). Moreover, low Cr levels may increase the risk of cardiovascular diseases (Thomas and Gropper, 1996). Chromium is not easily absorbed and shows low levels in the organism, explaining the absence of data on its toxicity. However, high doses of Cr have been related to chromosomal damage, alter-

ations in the kidney and liver, and metallic-mineral disorders (Guerrero-Romero and Rodríguez-Morán, 2005).

## 2.7 Cobalt

There are not many data on levels of cobalt (Co) in foods of plant origin in the scientific literature. The available data showed low levels of this micronutrient, often under 0.001 mg/100 g, with the lowest levels observed in vegetables (Szefer and Grembecka, 2007). The RDA for Co has been defined at around 300 micrograms (http://www.anyvitamins.com/rda.htm). Cobalt is required in the haematopoiesis of red blood cells and in preventing anaemia (Narasinga Rao, 2003). Its function is closely related to the physiological role of vitamin B12 in the production and maintenance of red blood cells. Moreover, Co stimulates appetite, and promotes growth and energy release (Kräutler, 2005; Mertz, 1981). Excessive intake of Co may damage the heart muscles, elevate the haemoglobin concentration, cause congestive heart failure and may cause damage to the thyroid gland, reducing its activity (Barceloux et al., 1999).

## 2.8 Copper

Low levels of copper (Cu) have been described in vegetables, ranging from 0.004 to 0.24 mg/100 g, except legumes, that can be up to 0.5 mg/100 g. Fruits contain small amounts of Cu, ranging from 0.01 to 0.24 mg/100 g (Szefer and Grembecka, 2007). The RDA of Cu ranges between 1.0 and 1.6 mg per day (http://www.anyvitamins.com/rda.htm). Copper primary functions are related to enzyme function including Phase-I detoxifying enzymes (i.e., the cytochrome C oxidase family of enzymes) (Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008). In addition, Cu is also necessary for the development of connective tissue and nerve coverings (myelin sheath) (Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008) and also participates in the Fe metabolism (Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005). Copper may be accumulated in the adult body (liver and brain) up to a limit of 80 mg (Guerrero-Romero and Rodríguez-Morán, 2005), supporting deficient dietary intake, without inducing clinical symptoms of toxicity for a short period of time. Cu deficiency is not frequent in humans, although it can cause several haematological symptoms such as normocytic, hypochromic anaemia, leucopenia and neuropenia, and skeletal disturbances (Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005). Toxic levels of Cu have been related to liver damage in chronic intoxication and gastrointestinal effects with cramps, nausea, diarrhoea and vomiting in acute episodes (Guerrero-Romero and Rodríguez-Morán, 2005).

## 2.9 Iron

Iron (Fe) contents in vegetables and fruits are low, varying from 0.13 to 3.01 mg/100 g. The iron in foods of plant origin is mostly present in the form of insoluble complexes of Fe<sup>3+</sup> with phytic acid, phosphates, oxalates and carbonates. However, the bioavailability of the Fe present in foods is less than 8%. Nuts and cocoa powder may be a good source of Fe (16.1 and 25.8 mg/100 g, respectively; Szefer and Grembecka, 2007; http://www.anyvitamins.com/rda.htm). The recommended intake of iron is 8–18 mg per day (http://

www.anyvitamins.com/rda.htm). The major function of Fe is related to the synthesis of haemoglobin and myoglobin (Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008). It is also required for energy production. The first reason for Fe deficiency is inadequate Fe intake (Lukaski, 2004). Severe Fe deficiency results in hypochromic anaemia (Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005). Toxic levels of Fe in the body may be a consequence of genetic or metabolic disorders, frequent blood transfusions or excessive intake. An excess of Fe over a long period could result in liver and heart damage, diabetes, and skin changes (Fraga and Oteiza, 2002).

### 2.10 Manganese

Fruits and vegetables are also characterised by a low content of manganese (Mn). Vegetables contain Mn in the range 0.01-0.078 mgh/100 g and fruits 0.01-0.66 mg/100 g (Szefer and Grembecka, 2007). The recommended intake of Mn is 2 mg/day (http:// www.anyvitamins.com/rda.htm), and its main physiological function is being an enzyme cofactor involved in antioxidant reactions related to the glucose metabolism (metabolism of carbohydrates and gluconeogenesis; Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008). Deficiencies in Mn are extremely rare but have shown a reduction in cholesterol, red blood cells and mucopolysaccharide abnormalities. Under experimental conditions signs of a scaly rash and low levels of plasma cholesterol have been observed (Shenkin, 2008). An excess of Mn produces a toxic effect in the brain, causing a Parkinson-like syndrome (Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008).

## 2.11 Molybdenum

Molybdenum (Mo) is present in plant-based foods, normally at low levels. However, certain foods may concentrate extremely high levels of Mo. The range of variation between foods is very wide (from  $1 \times 10^{-6}$  mg/100 g in wine to 0.15 mg/100 g in peas). Canned vegetables contain up to 0.03 mg/100 g (Szefer and Grembecka, 2007). Doses <250  $\mu$ g are considered safe (http://www.anyvitamins.com/rda.htm). Molybdenum function is related to the turnover of amino acids and purine metabolism, assisting in the elimination of secondary dangerous compounds (nitrosamines). Furthermore, Mo is a cofactor for oxidant enzymes, especially sulphite oxidase and xanthine oxidase (Shenkin, 2008). A Mo deficiency constitutes a hereditary metabolic disorder characterised by severe neurodegeneration, resulting in early childhood death (Schwartz, 2005). Toxic quantities and excess of Mo may interfere with the metabolism of Co and might give symptoms of anaemia and slow growth (Xiao-Yun

## 2.12 Nickel

et al., 2006).

Vegetables usually present nickel (Ni) levels in the range of  $5 \times 10^{-4}$  to 0.28 mg/100 g, and fruits between <0.004 and 0.05 mg/100 g (Szefer and Grembecka, 2007). The recommended daily intake of Ni is in the range of 302–735 µg (Roychowdhury et al., 2003).

#### 2.13 Selenium

The concentration of selenium (Se) ranges from  $10^{-4}$ to 0.06 mg/100 g in foods of plant origin (Szefer and Grembecka, 2007). Seventy micrograms per day is taken as the required dosage for this micronutrient (http://www.anyvitamins.com/rda.htm). Selenium is an essential component of selenoproteins, which are implicated in antioxidant reactions (Guerrero-Romero and Rodríguez-Morán, 2005). In addition, although Se functions are not fully known, it seems that it also presents activity related to thyroid and immune system functions through its intervention (Shenkin, 2008). Selenium is associated with marked reductions in risks of several types of cancer (Combs, 2004) and its deficiency may contribute to heart disease, hypothyroidism and deficiencies in the immune system (Guerrero-Romero and Rodríguez-Morán, 2005; Combs, 2000; Zimmermann and Köhrle, 2002). An excess of Se has been related to several symptoms including: gastrointestinal upset, hair loss, fatigue and mild nerve

damage. However, Se toxicity is not frequent and is related to accidental exposures (Guerrero-Romero and Rodríguez-Morán, 2005).

## 2.14 Zinc

The concentration of zinc (Zn) in plant-based foods generally varies from 0.05 to 11.8 mg/100 g. The lower levels of Zn are found in fresh fruits (0.02-0.61 mg/100 g). Fruit juices and beverages are characterised by low levels of Zn ranging from 0.01-0.27 mg/100 g (Szefer and Grembecka, 2007). Recommended daily Zn consumption ranges from 8-11 mg (http://www.anyvitamins.com/rda.htm; Lukaski, 2004). It is required for the structure and activity of more than 100 enzymes (Huskisson et al., 2007; Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008), for the synthesis of nucleic acids and proteins, for cellular differentiation, and for glucose use and insulin secretion (Lukaski, 2004). This mineral takes part in the Zn fingers associated with DNA, haemoglobin, myoglobin and cytochromes (Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008). The bioavailability of Zn is reduced by the presence of large amounts of other elements such as Fe or Cu (Shenkin, 2008). Zn deficiency is relatively frequent and well characterised, and the absence of Zn negatively affects the immune system efficacy, and the sensibility of taste and smell senses, and impairs DNA synthesis (Guerrero-Romero and Rodríguez-Morán, 2005; Shenkin, 2008). It has also been described that Zn deficiency produces hair loss and hypochromic anaemia (Shenkin, 2008). Zn toxicity shows both acute and chronic effects. Intakes of 150-450 mg per day over an extended period of time have been associated with poor Cu levels, altered Fe and immune functions, and reduced levels of HDL (Guerrero-Romero and Rodríguez-Morán, 2005; Hamilton et al., 2001).

## 3 Environmental Stress Affecting Plant Mineral Content

Abiotic stresses such as high salt levels, low water availability and extreme temperatures can severely modify the mineral and nutritive quality of the crops for human consumption. These types of external stresses are becoming increasingly important because of the global reduction in the availability of water resources of good quality for irrigation, which indeed is affecting the plant mineral status and consequently, the nutritional quality of a given cultivar. To date, the reports have mainly studied the influence of environmental stresses on the carbohydrates (sugars), amino acids or antioxidant production of vegetables and fruits, and most investigations have been focused on salinity as the main abiotic stress. However, there is limited information about the influence of general abiotic factors on the mineral content of plant-based foods and food products as a bioindication of the food nutritive value and quality. In general, the mineral nutrient contents change when external conditions affect the plant growth (i.e., environmental stress) and there is a reduced plant growth and reduced biomass at harvest, accompanied by less dilution of nutrients on a dry mass basis (Fig. 2).

## 3.1 Salinity

The use of saline water for irrigation may affect the mineral composition of plants and, therefore, the fruit quality. In a saline environment, ion homeostasis can be disturbed by excessive uptake of Na<sup>+</sup> and Cl<sup>-</sup>. Competition between these and further anions and cations has been well documented over the last 20 years (Sharpley et al., 1992; Lopez and Satti, 1996; De Pascale et al., 2005). Thus, in general, salinity reduces phosphate uptake and accumulation in crops as

well as Ca<sup>2+</sup> soil bioavailability and transport, which affects the quality of both vegetative and reproductive organs including fruits and edible parts of the plants. In addition,  $Na^+$  and  $Cl^-$  ions may reduce  $K^+$  and NO<sub>3</sub><sup>-</sup> uptake, respectively (Grattan and Grieve, 1999). Under saline stress, a reduction in  $NO_3^-$  content has been observed in edible florets of broccoli (Brassica oleracea var. italica) and in tomato (Solanum lycopersicon) (Lopez-Berenguer et al., 2009) and although the reports on the effect of nitrate on human health are still conflicting, its reduction in foods could add a nutritional value to the cultivar of interest (Anjana and Iqbal, 2007). Also, in these reports, concentrations of Na<sup>+</sup> and Cl<sup>-</sup> were higher in the leaves than in the florets, in agreement with the fact that under saline stress plants attempt to minimise the concentration of toxic ions in their reproductive organs (Hachicha et al., 2000). However, Del Amor et al. (2001) found that in tomato fruits, total anion Cl<sup>-</sup> and NO<sub>3</sub><sup>-</sup> concentrations increased by 11% as the salinity level increased from 2 to 8 dS·m<sup>-1</sup> but fruit K<sup>+</sup>, Na<sup>+</sup>, Ca<sup>2+</sup> and Mg<sup>2+</sup> contents were reduced significantly by salinity levels. Interactions between salinity and fertilisation have been described and concentrations of P, K<sup>+</sup>, Mg<sup>2+</sup>, Cu and Zn significantly decreased at high salinity and when urea was used as a nitrogen source. In this case, the total nitrogen concentration was not affected. Other studies on tomato and salinity showed that fruit Ca<sup>2+</sup> was also decreased by salinity or  $NH_4^+$ , with the negative effect of NH<sub>4</sub><sup>+</sup> being higher than the effect of salinity (Flores et al., 2003).

On the other hand, saline stress is a condition that may cause a combination of complex interactions affecting the plant metabolism or the inner nutritional



Fig. 2 Greenhouse experiment evaluating environmental stresses (salinity and drought) in horticultural crops

requirements, but little information on the distribution of essential minerals in plants for foods grown under salinity has yet been published. Moreover, the effects of salinity on mineral contents are often equivocal depending on the species or cultivar and the specific plant organ (De Pascale et al., 2005). On this subject, it has been reported that salinity can originate stimulatory as well as inhibitory effects on the uptake of some micronutrients by crop plants. Thus, as recently observed in two strawberry cultivars under salt stress, the mineral status of the berries was improved (increased Na<sup>+</sup> and Cl<sup>-</sup>, as well as N and P contents), but a different response was detected for K<sup>+</sup> and Zn, which remained unaffected in the less-sensitive cultivar, and rose in the sensitive cultivar (Keutgen and Pawelzik, 2008).

The uptake of Fe, Mn, Zn and Cu generally increases in crop plants under salinity stress (Alam, 1994). However, the detrimental effects of NaCl stress on the nutrition of bean plants reflected differences in distinct plant organs and showed higher concentrations of Cl<sup>-</sup> and Mn in roots, Cl<sup>-</sup>, Fe and Mn in leaves, and Cl<sup>-</sup> and Fe in fruits (Carbonell-Barrachina et al., 1998). Therefore, when applying moderate salinity levels for quality improvement, it is necessary to consider changes in the pool of mineral nutrients depending on the sensitivity of the cultivar and differences in mineral accumulation in the plant organs in order to avoid negative effects of the treatment.

In the fruits of courgette plants (Cucurbita pepo L. var. Moschata) an 80 mM NaCl treatment improved yield and fruit quality (Villora et al., 1999) and significantly increased the concentrations of micronutrients (Fe and Zn mainly) in the edible part of this crop (Víllora et al., 2000). In addition, the concentrations of total Mn and total extractable Fe, Cu and Zn followed no linear pattern in response to the increased NaCl concentrations. These results for courgette contrast with findings for tomato, squash and green beans, in which the level of each microelement reportedly fluctuates with salinity, perhaps due in part not only to the salt treatments but also largely to the type of crop and the cultivar used in each experiment (Grattan and Grieve, 1999). In a recent report,  $Ca^{2+}$ ,  $Mg^{2+}$ ,  $K^+$  and Mn contents decreased in the hot pepper fruits of a sensitive cultivar of Capsicum annuum L. as NaCl concentration increased (Ramirez-Serrano et al., 2008). Some studies on salinity in grafted plants of a "Star Ruby" grapefruit scion on two rootstocks, "Cleopatra" mandarin and "Carrizo" citrange, showed

that fruits from saline treatment on "Carrizo" had Cl<sup>-</sup> and Na<sup>+</sup> concentrations (2,87 and 1,6 times higher, respectively) than fruits from no saline treatments. Moreover, in the first harvesting, salinity increased K<sup>+</sup> concentration in the juice of fruits from trees grafted on "Carrizo" and treatments with 30 mM of NaCl decreased Ca<sup>2+</sup> concentration in fruits from trees grafted with both rootstocks. However, salinity had no major effect on juice K<sup>+</sup> concentration on the second harvesting date or on juice Mg<sup>2+</sup> concentration at both sampling times (García-Sánchez et al., 2003).

Plants respond to environmental stress by synthesising signalling molecules that activate a range of signal transduction pathways. Several such signalling molecules have been identified in plants such as Ca, jasmonic acid (JA), ethylene ( $C_2H_4$ ) and salicylic acid (SA) derivatives. However, the effect of signalling molecules applied externally under stressful conditions (saline treatments) on the plant mineral uptake is not fully understood or well documented. Positive effects of SA on the ion uptake, and inhibitory effects on Na<sup>+</sup> and Cl<sup>-</sup> uptake have been described for maize plants under salinity (Gunes et al., 2007). Similar effects of SA on the Na<sup>+</sup>, K<sup>+</sup>, Ca<sup>2+</sup> and Mg<sup>2+</sup> contents in wheat plants grown under salinity have been described (Al-Hakimi and Hamada, 2001).

In addition to its role as a cellular messenger, effects of Ca<sup>2+</sup> on integrity of membranes, rigidity of the cell wall, and maintenance of cell-to-cell contact are reported. Supplemental Ca<sup>2+</sup> has been successful in improving crop quality due to the correction of Ca<sup>2+</sup> deficiencies induced by Na<sup>+</sup>. Under osmotic stress, the distribution of  $Ca^{2+}$  to the distal end of fruits is decreased, leading to a local deficiency of Ca<sup>2+</sup> that causes rotting at the distal end of fruits known as blossom end rot (BER) (Ho et al., 1993; Saure, 2001; Guichard et al., 2001). However, BER is known to be affected not only by one factor, but also by interactions between water availability, salinity and nutrient ratios in the root zone, and the product of average daily solar radiation and air temperature, root temperature and air humidity (Adams and Ho, 1993; Ehret and Ho, 1986). In apples, it has also been observed that low Ca contents are associated with bitter pit disease (Fukumoto et al., 1987) or pit breakdown (Tomala and Dilley, 1990).

In general, salinity influences the uptake and transport of other ions by the plant and such antagonism could occur between Na<sup>+</sup> and Ca<sup>2+</sup>, K<sup>+</sup> or Mg<sup>2+</sup> and between Cl<sup>-</sup> and NO<sub>3</sub><sup>-</sup>. These effects may be involved in the occurrence of nutritional disorders in plant tissues, affecting food quality.

### 3.2 Drought

Limited water supply in many areas of the world, especially in arid and semiarid regions, is a major problem in irrigated agriculture. In recent years, it has become clear that the maintenance of a slight water deficit can improve the partitioning of carbohydrates to reproductive structures such as fruit and also control excessive vegetative growth (Chalmers et al., 1981). This is called "regulated deficit irrigation", consisting of irrigation input being removed or reduced for specific periods during the growth cycle of crops (Chalmers et al., 1986). This technique results in more efficient use of irrigation water and often improves product quality (Turner, 2001). Rouphael et al. (2008) showed that using three different levels of irrigation based on evapotranspiration (ET) rates (1.0, 0.75 and 0.5 ET) and two grafting treatments on watermelon plants no significant differences among treatments were observed for P and Ca<sup>2+</sup> concentrations, whereas K<sup>+</sup> and Mg<sup>2+</sup> concentrations were significantly improved by both the irrigation rate and grafting combination with no significant differences between irrigation and grafting interaction.

Drought and salinity can differentially affect the mineral nutrition of plants. While salinity may cause nutrient deficiencies or imbalances, due to the competition of Na<sup>+</sup> and Cl<sup>-</sup> with other nutrients such as K<sup>+</sup>, Ca<sup>2+</sup> and NO<sub>3</sub><sup>-</sup>, drought can affect nutrient uptake and impair translocation of some nutrients.

It has been recently reported that in banana, the main effect of drought was to reduce  $K^+$  levels, which is the major mineral nutrient in this fruit. By contrast, the content of certain elements increased (i.e.,  $Ca^{2+}$ ,  $Na^+$ , Fe and Zn), or remained stable (i.e.,  $N, P, Mg^{2+}$ , Mn and Cu) under the drought treatment, which also generated a positive effect on the organoleptic properties of the fruit (Mahouachi, 2007). After rehydration, the mineral content of the bananas was similar between stressed and non-stressed plants. These data illustrate the ability of this cultivar to maintain relatively normal levels of minerals and functional fruit tissues after dehydration despite the long period of water stress. Nonetheless, the fruits lost their commercial value to

a certain degree (reduced size and biomass) after the period of water stress.

Wild plants play an important role in the diet of inhabitants in different parts of the world. These plants tend to be drought-resistant and are gathered both in times of abundance and times of need, and for this reason numerous reports have been focused on wild edible plants. In a study of dietary practices in Northeastern Nigeria, it was observed that the edible wild species available during the wet season generally were inferior in micronutrient mineral contents compared with the dry season plants (Lockett et al., 2000). Commonly consumed species of edible wild barks, fruits, leaves, nuts, seeds, and tubers were analysed and Kuka bark (Adansonia dig*itata*), given to infants, was high in Ca<sup>2+</sup>, Cu, Fe and Zn. Cediya (Ficus thonningii), dorowa (Parkia biglobosa) and zogale (Moringa oleifera) were also good sources of Ca<sup>2+</sup>, Fe, Cu and Zn. Fruits, leaves and nuts of aduwa (Balanites aegyptiaca) are widely used during the dry and drought seasons. Tsamiya seeds (Tamarindus indica), consumed commonly during pregnancy, were good sources of Zn, and Kirya seeds (Prosopis africana) contained the highest Zn concentrations. Shiwaka leaves (Veronia colorate), consumed by pregnant women to increase breastmilk production, were high in P,  $Mg^{2+}$  and  $Ca^{2+}$ .

In another report it has been illustrated that the mineral content of some edible wild leaves contained higher N,  $K^+$ ,  $Ca^{2+}$  and  $Mg^{2+}$  concentrations than those of some commonly used vegetables such as spinach (*Spinacia oleracea*), pepper (*Capsicum annum*), lettuce (*Lactuca* spp.) and cabbages (*Brassica oleracea*). However, P, S and Na<sup>+</sup> contents were lower, and Fe, Mn, Zn and Cu levels were equal (Turan et al., 2003).

The time of application as well as the duration of water stress during the fruit development can influence the mineral content. In 'Williams' pears a reduced water supply at the end of the fruit development (late water stress) caused smaller reductions in the uptake of  $Ca^{2+}$ ,  $K^+$  and B within the fruits, while at the beginning of fruit development (early water stress) only a lower content of  $Ca^{2+}$  was observed in the fruits. Reduced water supply either at the beginning or the end of the productive cycle induced a higher N content in the fruits. In the treatment of early water stress the K contents were higher than in the untreated controls (Hudina and Stampar, 2000).

The potential of roots to absorb nutrients generally declines in water-stressed plants, presumably due to a decline in nutrient element demand, but the ability to take up and transport the mineral nutrients differs in distinct crops and depends on the plant's tolerance to drought.

## 3.3 Extreme Temperatures

It is evident that the roots play a principal role in the absorption of the mineral elements that will be translocated to the aerial parts of the plant. In general, a consistent decrease in the nutrient concentrations in the plant shoots is parallel to the growth suppression at low root temperatures. Growth and mineral composition of fruits in potted trees were studied at two temperatures (19 and 24°C) in 'Golden Delicious' and in 'Cox's Orange Pippin' apples (Malus domestica) (Tromp, 1975). In this report the levels of K, N, Mg and P were increased at the higher temperature. With respect to Ca, the high temperature regime reduced its influx in 'Golden Delicious' but favoured the Ca influx in the 'Cox's Orange Pippin' apples. Similarly, in two cultivars of tomato with contrasting response to elevated temperature, Ca was poorly transported to the fruits but in fruit explants, the elevated temperature (40°C) increased the Ca import into the fruits in both cultivars. This permanent flux of Ca to the fruits may have a pivotal role in maintaining an optimal level of  $Ca^{2+}$  in the cytoplasm of fruit cells, as a factor for increasing the tolerance to high temperatures (Starck et al., 1994). On the other hand, only a few studies have examined the effect of differences in temperature regimes between day and night on the mineral status of fruits or vegetable foods, but low concentration of nitrate was observed as a consequence of variations in temperature in root and shoot Ca<sup>2+</sup> due to lower night air temperatures (Gent and Ma, 2000).

## 3.4 Light Intensity

A certain influence of the light on the transport of nutrients from shoots to the fruit through the transpiration stream has been suggested in different studies. Caruso et al. (2004) reported that shading caused a reduction in the content of the main mineral elements except for nitrates in strawberry fruits. Also, the effects of sunlight on the mineral contents of apples were investigated (Iwane and Bessho, 2006). The treatment of sunlight from East and West directions on 75 apples from the inside and outside of the crown of the trees revealed a significant negative correlation between the amount of solar radiation received by the fruit and its mineral content for  $K^+$ ,  $Ca^{2+}$  and Zn. The concentration of minerals in apples grown on the inside was higher than in the fruits grown outside of the crown. The concentration of minerals was higher in apples grown on the shaded side than in the apples grown on the sunny side. No significant differences in mineral concentrations were observed between fruit grown on the East or West orientation.

Effects of exposure to light and air movement on the accumulation of some mineral elements in fruits of kiwifruit (*Actinidia deliciosa* var. deliciosa) have been presented (Montanaro et al., 2006), where the main differences were found for  $Ca^{2+}$ , exhibiting twice the content in exposed fruits (>40% full sunlight) than in shaded fruit (<20% full sunlight).

Light is one of the main external factors influencing the nitrate concentrations in vegetables. Several human health hazards due to nitrate toxicity have been identified. The accumulation of nitrate in the plant tissues is more frequent under poor light conditions in leafy vegetables such as lettuce, spinach or kohlrabi (*Brassica oleracea* var. gongylodes L.) (Blom-Zandra and Lampe, 1985; Steingröver et al., 1986; Sritharan and Lenz, 1992). A controlled nutritional regime is then needed to reduce the leaf nitrate in the leaves (i.e., lettuce) under such unfavourable light conditions (Demsar et al., 2004).

Anjana et al. (2006) have reported that nitrate concentration was lowest at noon on a sunny day in spinach leaves. Thus, the time at which plants contain the lowest nitrate concentration may vary with the environmental conditions in different geographical regions of the world, and also depends on the interaction with other environmental factors. Santamaria et al. (1999) observed in different fresh vegetables that under conditions of low light availability, an increase in temperature increases the nitrate accumulation.

## 4 Fertilisation Practices and Mineral Content in Food Crops

The supply of essential nutrients for the health of consumers by improving the fertilising practices in the productive sector has awakened great interest in recent years. Here, we will summarise the available information on the effects of fertiliser applications on the mineral content of crops in relation to food quality for human consumption.

## 4.1 Nitrogen Fertilisation

The effect of N fertilisers on the mineral content of edible parts or fruits and vegetables is variable depending on the doses applied, the nutrient analysed, the species under study, and the organ to be consumed. Thus, in tomato grown under different N doses (0, 60, 120 and  $180 \text{ hg ha}^{-1} \text{ N}$ ), only the higher N doses increased total fruit N levels but, antagonistically, K<sup>+</sup> levels decreased continuously with the increased N (Cserni et al., 2008). In tuber and root crops such as potato (Solanum tuberosum L.) and sweet potato (Ipomoea batatas L.), which have enlarged underground stems and roots as edible parts, the application of N fertilisers usually led to increased tuber N concentration (Eppendorfer and Eggum, 1994). Also, the content of elements such as  $K^+$ , P, Ca<sup>2+</sup> and Mg<sup>2+</sup> in mature tubers of potato was not significantly different to the untreated tubers after N fertilisation  $(0-200 \text{ kg.ha}^{-1})$  (Ilin et al., 2002). In contrast, in broccoli sprouts (Brassica oleracea var. Italica), higher S concentrations were found when increasing N and S fertilisation rates up to a determined dose of N and S, but higher N rates did not yield higher S uptake (Aires et al., 2007).

In experiments studying the influence of the  $NH_4^+/NO_3^-$  ratio in the nutrient solution and its effects on mineral status (in *Brassica oleracea* var. Italica), when this ratio was 0.5:0.5, the concentrations of P,  $K^+$ ,  $Ca^{2+}$  and  $Mg^{2+}$  were all higher in cabbage roots and leaves than those in plants grown in nutrient solutions with different ratios of decreased  $NH_4^+$  supply (Zhang et al., 2007).

It is well known that the application of N improves plant growth and yield. However, the application of high concentrations of N not only contaminates the environment, but also causes  $NO_3^-$  accumulation in the leaves of vegetable crops, which have been found to be the major source of toxic  $NO_3^-$  uptake by humans. In general, vegetables that are consumed with their roots, stems and leaves have a high  $NO_3^-$  accumulation, whereas those with only fruits as the edible part have low  $NO_3^-$  accumulation rates (Zhou et al., 2000). In addition, NO<sub>3</sub><sup>-</sup> contents vary depending on the organ of the plant (Santamaria et al., 1999; Anjana et al., 2006) and the physiological age of the plant (Maynard et al., 1976; Anjana et al., 2006). Field experiments have shown that  $NO_3^-$  concentrations in leafy vegetables were positively correlated with N rates, and N fertiliser added to the soil was the major cause of  $NO_3^-$  accumulation in vegetables (Wang et al., 2001). In the same way, Zhang et al. (2007), and Staugaitis et al. (2008) found a linear increase in  $NO_3^-$  concentration with increasing N fertilisation in cabbage (Brassica campestris) leaves and roots, and heads, respectively.

## 4.2 Phosphorus Fertilisation

Phosphorus promotes root growth, enhances nutrient and water-use efficiency, and increases yield. Therefore, since absorption and reduction of  $NO_3^-$  is a process which consumes ATP, the metabolism of  $NO_3^-$  is related to P supply. In pot experiments, it has been observed that a high soil N:P ratio was one of the key causes of  $NO_3^-$  accumulation in vegetables (Gao et al.,1989). In contrast, P fertilisation decreased  $NO_3^$ concentration in cabbage (*Brassica campestris*) and canola (*Brassica napus*), and had no significant effect in spinach (*Spinacia oleracea*) (Wang and Li, 2004).

Togay et al. (2008) studied the effect of different doses of P (10, 40 and 80 kg ha<sup>-1</sup>) on the P content in grain of dry bean (*Phaseolus vulgaris*) and observed increased P levels when 80 kg ha<sup>-1</sup> P was applied. When looking at the micronutrients, Moreno et al. (2003) found that Fe and Mn concentrations in cucumber were higher in P-fertilised treatments compared with the unfertilised control. This effect is similar to what was found with edible fruits such as apple (*Malus sylvestris* Mill var. *domestica* (Borkh.)) where increased fruit P was obtained with P treatments. On the contrary, in cereals Komljenovic et al. (2006) revealed that maize grain (*Zea mays*) was less dependent on P fertilisation compared with the leaf; and it has also been shown in pear (*Pyrus communis*) that foliar P fertilisation decreased the content of B and Zn in fruits (Hudina and Stampar, 2002).

## 4.3 Potassium Fertilisation

Potassium is closely related to N assimilation in plants and can accelerate transport of  $NO_3^-$  from roots to aboveground plant parts. Thus, Zhou et al. (1989) showed that compared with the control,  $NO_3^-$  concentration in cabbage (*Brassica campestris*) decreased with the application of K<sup>+</sup>, whereas it increased in spinach (*Spinacia oleracea*) (Gao et al., 1989).

In relation to other minerals,  $K^+$  fertilisation has different effects. In fact, Hudina and Stampar (2002) showed that foliar fertilisation with  $K^+$  increased the content of K in pears (*Pyrus communis* L.), whereas  $K^+$  concentrations in broccoli heads (*Brassica oleracea* var. Italica) showed no differences among four levels of  $K^+$  fertilisation (Vidal-Martínez et al., 2006). In other studies, the concentrations of B and Zn in pears (Hudina and Stampar, 2002) as well as N concentrations in potato (*Solanum tuberosum*) and sweet cabbage (Eppendorfer and Eggum, 1994) decreased with increased  $K^+$  fertilisation.

## 4.4 Sulphur Fertilisation

Sulphur fertilisation may be recommended for certain crops to reduce the undesirable  $NO_3^-$  contained in their edible parts. In fact, an increased soil S level significantly reduced  $NO_3^-$  concentrations in tubers and leaves of kohlrabi (Losak et al., 2008), and turnip tops (*Brassica rapa* L.) (De Pascale et al., 2007). In S-deficient soils, the application of S fertilisers can decrease the tuber N concentration in potato (*Solanum tuberosum*) due to increased dry mass yield (Eppendorfer and Eggum, 1994). Nevertheless, in a greenhouse pot experiment using 'Luna' kohlrabi (*Brassica oleracea*), the effect of S fertilisation on N content in tubers and leaves was insignificant (Losak et al., 2008).

#### 4.5 Calcium Fertilisation

Leafy vegetables can be an excellent dietary source of calcium, and are a good alternative for individuals with a diet low in dairy products. Increasing the calcium content in leafy vegetables through fertilisation management could further improve their nutritional benefits. Thus, it has been observed that in lettuce produced in a hydroponic system an increase in  $Ca^{2+}$  concentration in the nutrient solution increased the  $Ca^{2+}$  levels in the leaves (Neeser et al., 2007).

It has also been shown that  $Ca^{2+}$  application increased  $Ca^{2+}$  concentrations in peripheral layers of apple fruits and reduced K<sup>+</sup> concentrations (Grimm-Wetzel and Schonherr, 2007). Similar results were reported by Val et al. (2008), where  $Ca^{2+}$  treatments increased the concentration of  $Ca^{2+}$  in the skin, but not in the flesh of fruit, and several sprays were needed to promote a prolonged increase in the concentration of  $Ca^{2+}$  in the skin. However,  $Ca^{2+}$  sprays did not influence the concentrations of  $Mg^{2+}$  and K<sup>+</sup>. In a recent study the application of  $CaCl_2$  increased the  $Ca^{2+}$  content in litchi fruit; firmness and skin colour were affected, and some positive correlations with leaf and fruit K<sup>+</sup> were detected (Cronje et al., 2009).

In kiwifruit, fruit quality is associated with the correct  $Ca^{2+}$  level; however, the application of a biostimulant such as  $Ca^{2+}$  fertiliser, which is recommended to prevent calcium deficiency resulting from lack of uptake into fruit, did not affect fruit yield (Otero et al., 2007).

#### 4.6 Microelement Fertilisation

Although the majority of experiments have been performed using the widely distributed N, P and K<sup>+</sup> fertilisers, there are also a few pieces of information about the effects of microelement fertilisers on the edible parts of plants for human consumption (Baize et al., 2009). Graham et al. (2001) showed that application of Zn fertiliser to Zn-deficient soil at sowing significantly increased the Zn concentration in wheat grain. Also, the content of Zn and several other micronutrients, such as I, Se, Cu and Ni, was usually enhanced by application of the appropriate mineral forms (Wang et al., 2008). Micronutrient foliar fertilisation seems to be a cheap and effective method, depending on the nutrient supplied and the time of application. In this way, Wojcik and Wojcik (2003) showed that foliar B sprays before full bloom or after harvest increased B concentrations in fruitlets of pear (*Pyrus communis* L.) at 40 days after flowering.

Selenium deficiency is a very serious nutritional and health problem. That is why the effect of selenite and selenate fertilisation on Se content has been widely studied (Chen et al., 2002; Fang et al., 2008; Ducsay et al., 2009). In reported experiments, selected mineral contents were higher with application of selenate than selenite to certain species. Thus, a higher content of Se in rice (Oryza sativa) (Chen et al., 2002) and S in lettuce plants (Lactuca sativa cv. Philipus) (Ríos et al., 2008) were found. Other studies showed that differentiated doses of selenite in soil caused a significant increase in Se content in dry matter of wheat grain (Ducsay et al., 2009), whereas non-significant effects were observed in lettuce (Ríos et al., 2008). In addition, Fang et al. (2008) indicated that Zn and Se were the main variables increasing the Zn, Se and Fe contents of rice. Therefore, the application of Zn, Se and Fe mixed fertiliser as foliar spray could alleviate the physiological deficiency of these micronutrients in rice.

## 4.7 Organic Farming Versus Mineral Fertilisation

The massive use of chemical fertilisers in intensive agriculture has greatly increased concern for the declining fertility of soils. Soil nutrient depletion is the result of increasing pressure on agricultural land (Wopereis et al., 2006; Lal, 2009). That is why organic inputs are required to ensure that intensive systems do not threaten the sustainability of land use. However, small farmers are reluctant to use organic wastes or composts because organic fertilisers do not release nutrients as fast as mineral fertilisers and they do not supply a balanced ratio of nutrients at the right time (Bath, 2000; Kirchmann et al., 2002; Gunnarsson, 2003). However, demand for organically-grown products has risen steadily and the number of growers adopting organic farming systems has also increased, because organic foods are believed to be more nutritious than conventionally-grown foods, with a better balance of vitamins and minerals.

Nevertheless, the scientific community has not conclusively shown that organic products are more nutritious than conventionally-grown foods (Winter and Davis, 2006). Thus, it seems to be important to see how organic and inorganic fertilisers affect food quality in terms of mineral content. Also, the negative effects that fertilisers have on food quality must be considered, since the anthropogenic activities aimed at enhancing food production may facilitate the accumulation of undesirable substances.

Several experiments have been performed in order to compare the effects of organic and conventional (mineral) fertilisers on the crop yield and nutritional status of plants, since organic yields are often lower compared with conventional production (Mäder et al., 2002; Dumas et al., 2003; Gopinath et al., 2008).

Usually, the organic-amended soils showed significantly higher soil mineral content (Edmeades, 2003). However, other authors indicated lower mineral contents for organically fertilised soils (Gosling and Shepherd, 2005).

The influence of organic soil fertilisation on nutrient content in crops has been studied and different results have been recorded. Some authors showed that the application of organic amendments improved the soil nutrient content, but did not always increase the plant nutrient concentration (Roe, 1998; Warman, 2005) since it depends on the crop type, the nutrient used, the climate parameters and the year of the study (Warman and Havard, 1997, 1998; Maqueda et al., 2001). Furthermore, the available scientific literature shows that some of the comparisons are not experimentally valid due to variation in crop varieties, timing in fertilisation, and handling and storage after harvesting (Warman and Havard, 1997). However, there are certain results that support that higher P and K<sup>+</sup> contents in wheat grain were obtained by applying organic amendments than the elemental contents using mineral fertilisers instead (Colla et al., 2002; Wszelaki et al., 2005; Gopinath et al., 2008; Basu et al., 2008).

The long-term use of organic composts (vegetal compost and green residue of previous crops) on greenhouse soils induced few differences in the macronutrient concentrations in the edible parts of food crops compared with the experiments using mineral fertilisation, although there was a trend of showing higher N concentration in minerally-grown crops and higher K<sup>+</sup> concentration in organically-grown crops (Herencia et al., 2007). Moreover, the NO<sub>3</sub><sup>-</sup>

concentrations in the edible parts of organically-grown crops were significantly lower than in the minerallyfertilised plots (Vogtmann et al., 1993; Williams, 2002; Malmauret et al., 2002; Hajslova et al., 2005). This can provide a clear benefit for human health. Nevertheless, the results were variable depending on the crop, season cycle and year, and these factors must be considered carefully in the conclusions and potential recommendation to producers and consumers.

## 5 Conclusion

The quality of edible fruits concerning mineral contents may vary depending on interactions between cultivars, environmental factors such as light and temperature, composition of the nutrient solution, crop management practices, and the interaction of all these factors. This is the reason why all of them must be taken into account in order to characterise the nutritional value (mineral status) of fruits and vegetables, as well as the factors influencing the content of a specific element in a given cultivar. In addition, the physiological parameters of the fruit (stage of development, ripening, marketable maturity, physiological maturity, senescence) and the plant or tree as a whole are also of interest.

The influence that fertilisation practices may have on the mineral status and nutritive value of fruits and vegetables also depends on the fertiliser used, the macro- (i.e., N, P, K, S, etc.) or micronutrient (i.e., Fe, Mn, Zn, Cu, Ni, Co, Se, etc.) studied, and the plant part of interest for consumption (i.e., leaf, root, tuber, fruit). Similarly, a determined or programmed pattern cannot be established for the use of irrigation regimes with waters of different qualities. Nonetheless, fertilisation seems to remain one of the most practical and effective ways to control and improve the nutritional value of crops to meet the needs of the population, as well as proper water management integrating practices for food quality and safety. A large body of research results has been performed in the past decades on the effects of distinct agronomical practices on specific crops of human interest, but more concise and precise studies are needed to improve the load of essential microelements in foods and to prevent or avoid the accumulation of toxic or undesirable contaminants.

Acknowledgements This work was funded by the CI-CYT (AGL2006-06499) and the Fundación Seneca-Comunidad Autónoma de la Región de Murcia (08753/PI/08). This work was funded partly by a Convenio de Desempeño-UTA-MECESUP2 (Arica-Chile). E. Bastías received a fellowship from the Fundacion Séneca, 09769/IV2/08.

#### References

- Adams P., Ho L.C. (1993) Effects of environment on the uptake and distribution of calcium in tomato and on the incidence of blossom-end rot, Plant Soil 154, 127–132.
- Agrawal V., Agarwal M., Joshi S.R., Ghosh A.K. (2008) Hyponatremia and hypernatremia: Disorders of water balance, J. Assoc. Physicians India 56, 956–964.
- Ahokas R.A., Sun Y., Bhattacharya S.K., Gerling I.C., Weber K.T. (2005) Aldosteronism and a proinflammatory vascular phenotype: Role of Mg<sup>2+</sup>, Ca<sup>2+</sup>, and H<sub>2</sub>O<sub>2</sub> in peripheral blood mononuclear cells, Circulation 111, 51–57.
- Aires A., Rosa E., Carvalho R., Haneklaus S., Schnug E. (2007) Influence of nitrogen and sulfur fertilization on the mineral composition of broccoli sprouts, J. Plant Nutr. 30, 1035–1046.
- Alam S.M. (1994) Nutrient uptake by plants under stress conditions, in: Pessarakli M. (Ed.), Handbook of plant and crop stress, Marcel Dekker, New York, pp. 227–246.
- Al-Hakimi A.M.A., Hamada A.M. (2001) Counteraction of salinity stress on wheat plants by grain soaking in ascorbic acid, thiamin or sodium salicylate, Biol. Plant. 44, 253–261.
- Anjana U.S., Iqbal M. (2007) Nitrate accumulation in plants, factors affecting the process, and human health implications. A review, Agron. Sustain. Dev. 27, 45–57.
- Anjana U.S., Iqbal M., Abrol Y.P. (2006) Are nitrate concentrations in leafy vegetables within safe limits? Proceedings of the Workshop on Nitrogen in Environment, Industry and Agriculture, New Delhi, India, pp. 81–84.
- Baize D., Bellanger L., Tomassone R. (2009) Relationships between concentrations of trace metals in wheat grains and soil, Agron. Sustain. Dev. 29, 297–312.
- Barceloux D.G. (1999) Cobalt, J. Toxicol. Clin. Toxicol. 37, 201–216.
- Basu M., Bhadoria P.B.S., Mahapatra S.C. (2008) Growth, nitrogen fixation, yield and kernel quality of peanut in response to lime, organic and inorganic fertilizer levels, Bioresource Technol. 99, 4675–4683.
- Bath B. (2000) Matching the availability of N mineralisation from green-manure crops with the N-demand of field vegetables, Ph.D. Thesis, Swedish University of Agricultural Sciences, Uppsala, Sweden.
- Blom-Zandra M., Lampe J.E.M. (1985). The role of nitrate in the osmoregulation of lettuce (Latuca sativa L.) grown at different light intensities, J. Exp. Bot. 36, 1043–1052.
- Bo S., Pisu E. (2008) Role of dietary magnesium in cardiovascular disease prevention, insulin sensitivity and diabetes, Curr. Opin. Lipidol. 19, 50–56.
- Carbonell-Barrachina A.A., Burlo F., Mataix J. (1998) Response of bean micronutrient to arsenic and salinity, J. Plant Nutr. 21, 1287–1299.

- Caruso G. Villari A., Villari G. (2004) Quality characteristics of "Fragaria vesca L." fruits influenced by NFT solution EC and shading, Acta Hort. 648, 167–174.
- Chalmers D.J., Burge G., Jerie P.H., Mitchell P.D. (1986) The mechanism of regulation of Bartlett pear fruit and vegetative growth by irrigation withholding and regulated deficit irrigation, J. Am. Soc. Hortic. Sci. 111, 904–907.
- Chalmers D.J., Mitchell P.D., Vanheek L. (1981) Control of peach tree growth and productivity by regulated water supply, tree density and summer pruning. J. Am. Soc. Hortic. Sci. 106, 307–312.
- Chen L., Yang F., Xu J., Hu Y., Hu Q., Zhang Y., Pan G. (2002) Determination of selenium concentration of rice in china and effect of fertilization of selenite and selenate on selenium content of rice, J. Agr. Food Chem. 50, 5128–5130.
- Colla G., Mitchell J.P., Poudel D.D., Temple S.R. (2002) Changes of tomato yield and fruit elemental composition in conventional, low input, and organic systems, J. Sustain. Agr. 20, 53–67.
- Combs G.F. Jr. (2000) Food system-based approaches to improving micronutrient nutrition: The case for selenium, Biofactors 12, 39–43.
- Combs G.F. Jr. (2004) Status of selenium in prostate cancer prevention, Br. J. Cancer 91, 195–199.
- Cronje R.B., Sivakumar D., Mostert P.G., Korsten L.(2009) Effect of different preharvest treatment regimes on fruit quality of litchi cultivar "Maritius", J. Plant Nutr. 32, 19–29.
- Cserni I., Peto J.B., Huvely A., Nemeth T., Vegh K.R. (2008) Nitrogen, phosphorous, potassium, acid, sugar and vitamin c content in tomato grown in different soil types and under different nitrogen doses, Cereal Res. Commun. 36, 1415–1418.
- De Pascale, S., Maggio, A., Barbieri, G. (2005) Soil salinization affects growth, yield and mineral composition of cauliflower and broccoli, Eur. J. Agron. 23, 254–264.
- De Pascale S., Maggio A., Fogliano V., Ambrosino P., Ritieni A. (2001) Irrigation with saline water improves carotenoids content and antioxidant activity of tomato, J. Hortic. Sci. Biotech. 76, 447–453.
- De Pascale S., Maggio A., Pernice R., Fogliano V., Barbieri G. (2007) Sulphur fertilization may improve the nutritional value of *Brassica rapa* L. subsp *sylvestris*, Eur. J. Agron. 26, 418–424.
- Del Amor F.M., Martínez V., Cerdá A. (2001) Salt tolerance of tomato plants as affected by stage of plant development, Hortscience 37, 1260–1263.
- Demsar J., Osvald J., Vodnik D. (2004) The effect of lightdependent application of nitrate on the growth of aeroponically grown lettuce (*Lactuca sativa* L.), J. Am. Soc. Hortic. Sci. 129, 570–575.
- Ducsay L., Lozek O., Varga L. (2009) The influence of selenium soil application on its content in spring wheat, Plant Soil Environ. 55, 80–84.
- Dumas Y., Dadomo M., Di Lucca G., Grolier P. (2003) Effects of environmental factors and agricultural techniques on antioxidant content of tomatoes, J. Sci. Food Agr. 83, 369–382.
- Durlach J., Bac P., Durlach V., Rayssiguier Y., Bara M., Guiet-Bara A. (1998) Magnesium status and ageing: An update, Magnesium Res. 11, 25–42.
- Edmeades D.C. (2003) The long-term effects of manures and fertilisers on soil productivity and quality: A review, Nutr. Cycl. Agroecosyst. 66, 165–180.

- Ehret D.L., Ho L.C. (1986) Effects of osmotic potential in nutrient solution on diurnal growth of tomato fruit, J. Exp. Bot. 37, 1294–1302.
- Eppendorfer W.H., Eggum B.O. (1994) Dietary fibre, starch, amino acids and nutritive value of potatoes as affected by sulfur, nitrogen, phosphorus, potassium, calcium and water stress, Acta Agr. Scand. B Soil Plant Sci. 44, 107–115.
- Fang Y., Wang L., Xin Z., Zhao L., An X., Hu Q. (2008) Effect of foliar application of zinc, selenium, and iron fertilizers on nutrients concentration and yield of rice grain in china, J. Agr. Food Chem. 56, 2079–2084.
- Flores P., Navarro J.M., Carvajal M., Cerdá A., Martínez, V. (2003) Tomato yield and quality as affected nitrogen source and salinity, Agronomie 23, 1–8.
- Fraga C.G., Oteiza P.I. (2002) Iron toxicity and antioxidant nutrients, Toxicology 180, 23–32.
- French R.J., Zamponi G.W. (2005) Voltage-gated sodium and calcium channels in nerve, muscle, and heart, IEEE T. Nanobiosci. 4, 58–69.
- Fukumoto M., Nagai K., Yoshioka H., Aoba K. (1987) Mechanism of the development of a calcium-related disorder (bitter pit) in apple, Jpn Agric. Res. Q. 20, 248–252.
- Gao Z.M., Zhang Y.D., Zhang D.Y., Shi R.H., Zhang M.F. (1989) Effects of N, P and K on the accumulation of nitrate and the activity of nitrate reductase and superoxidase in leafy vegetables, Acta Hort. Sin. 16, 293–298.
- García-Sánchez F., Carvajal M., Cerdá A., Martínez V. (2003) Response of "Star Ruby" grape fruit on two rootstocks to NaCl salinity, J. Hortic. Sci. Biotech. 78, 859–865.
- Gent M.P.N., Ma Y.Z. (2000) Mineral nutrition of tomato under diurnal temperature variation of root and shoot, Crop Sci. 40, 1629–1636.
- Ghosh A.K., Joshi S.R. (2008) Disorders of calcium, phosphorus and magnesium metabolism, J. Assoc. Physicians India 56, 613–621.
- Gopinath K.A., Saha S., Mina B.L., Pande H., Kundu S., Gupta H.S. (2008) Influence of organic amendments on growth, yield and quality of wheat and on soil properties during transition to organic production, Nutr. Cycl. Agroecosys. 82, 51–60.
- Gosling P., Shepherd M. (2005) Long-term changes in soil fertility in organic arable farming systems in England, with particular reference to phosphorus and potassium, Agr. Ecosyst. Environ. 105, 425–432.
- Graham R.D., Welch R.M., Bouis H.E. (2001) Addressing micronutrient malnutrition through enhancing the nutritional quality of staple foods: principles, perspectives and knowledge gaps, Adv. Agron. 70, 77–142.
- Grattan S.R., Grieve C.M. (1999) Mineral nutrient acquisition and response by plants grown in saline environments, in: Pessarakli M. (Ed.), Handbook of Plant and Crop Stress, Marcel Dekker, New York, pp. 203–226.
- Grimm-Wetzel P., Schonherr J. (2007) Applications of calcium chloride to apple fruits increase calcium and reduce potassium concentrations in peripheral layers of fruits, Erwerbs-Obstbau 49, 75–83.
- Grusak M.A., Cakmak I. (2005) Methods to improve the crop delivery of minerals to human and livestock, in: Broadley M.R., White P.J. (Eds.), Plant Nutritional Genomics, Blackwel, pp. 265–286.

- Guerrero-Romero F., Rodríguez-Morán M. (2005) Complementary therapies for diabetes: The case for chromium, magnesium, and antioxidants, Arch. Med. Res. 36, 250–257.
- Guichard S., Bertin N., Leonardi C., Gary C. (2001) Tomato fruit quality in relation to water and carbon fluxes, Agronomie 21, 385–392.
- Gunes A., Inal A., Alpaslan M., Eraslan F., Bagci E.G., Cicek N. (2007) Salicylic acid induced changes on some physiological parameters symptomatic for oxidative stress and mineral nutrition in maize (*Zea mays* L.) grown under salinity, J. Plant Phys. 164, 728–736.
- Gunnarsson S. (2003) Optimisation of N release Influence of plant material chemical composition on C and N mineralization, Ph.D. Thesis, Swedish University of Agricultural Sciences, Uppsala, Sweden.
- Hachicha M., Cheverry C., Mhiri A. (2000) The impact of longterm irrigation on changes of ground water level and soil salinity in northern Tunisia, Arid. Soil Res. Rehab. 14, 175–182.
- Hajslova J., Schulzova V., Slanina P., Janné K., Hellenäs K.E., Andersson C. (2005) Quality of organically and conventionally grown patatoes: Four year study of micronutrients, metals, secondary metabolites, enzymic browning and organoleptic properties, Food Addit. Contam. 19, 524–532.
- Hall J.E. (2003) The kidney, hypertension, and obesity, Hypertension 41, 625–633.
- Hall J.E., Brands M.W., Henegar J.R. (1999). Mechanisms of hypertension and kidney disease in obesity, Ann. N.Y. Acad. Sci. 892, 91–107.
- Hamilton I.M.J., Gilmore W.S., Strain J.J. (2001) Marginal copper deficiency and atherosclerosis, Biol. Trace Elem. Res. 78, 179–189.
- Herencia J.F., Ruiz-Porras J.C., Melero S., Garcia-Galavis P.A., Morillo E., Maqueda C. (2007) Comparison between organic and mineral fertilization for soil fertility levels, crop macronutrient concentrations, and yield, Agron. J. 99, 973–983.
- Ho L.C., Belda R., Brown M., Andrews J., Adams P. (1993) Uptake and transport of calcium and the possible causes of blossom-end rot in tomato, J. Exp. Bot. 44, 509–518.
- Hudina M., Stampar F. (2000) Influence of water regimes and mineral contents in soil upon the contents of minerals, sugars and organic acids in pear fruits (*Pyrus communis* L. Cv. 'Williams'), Phyton-Ann. Rei. Bot. 40, 91–96.
- Hudina M., Stampar F. (2002) Effect of phosphorus and potassium foliar fertilization on fruit quality of pears, in: Tagliavini M. et al.(Eds.), Proceedings of the international symposium on foliar nutrition of perennial fruit plants, Acta Hort. 594, 487–493.
- Huskisson E., Maggini S., Ruf M. (2007) The role of vitamins and minerals in energy metabolism and well-being, J. Int. Med. Res. 35, 277–289.
- Ilin Z., Durovka M., Markovic V. (2002) Effect of irrigation and mineral nutrition on the quality of potato, in: Paroussi G., Voyiatzis D., Paroussis E. (Eds.), Proceedings of the second balkan symposium on vegetables and potatoes, Acta Hort. 579, 625–629.
- Iwane A., Bessho H. (2006) Factors affecting the variation of mineral components of apples part VI - Effects of sunlight

on the mineral content of apple fruit, J. Jpn Soc. Food Sci. 53, 151–157.

- Kahn T. (2008) Hypernatraemia, Nephrol. Dial. Transplant. 23, 4080–4081.
- Keutgen A.J., Pawelzik E. (2008) Quality and nutritional value of strawberry fruit under long term salt stress, Food Chem. 107, 1413–1420.
- Killilea D.W., Maier J.A.M. (2008) A connection between magnesium deficiency and aging: New insights from cellular studies, Magnesium Res. 21, 77–82.
- Kirchmann H., Johnston A.E.J., Bergström L.F. (2002) Possibilities for reducing nitrate leaching from agricultural land, Ambio 31, 404–408.
- Ko E.A., Han J., Jung I.D., Park W.S. (2008) Physiological roles of K+ channels in vascular smooth muscle cells, J. Smooth Muscle Res. 44, 65–81.
- Komljenovic I., Markovic M., Todorovic J., Cvijovic M. (2006) Influences of fertilization by phosphorus on yield and nutritional status of maize in Potkozarje area, Cereal Res. Commun. 34, 549–552.
- Kräutler B. (2005) Vitamin B12: Chemistry and biochemistry, Biochem. Soc. Trans. 33, 806–810.
- Lachance P.A. (1998) International perspective: Basis, need, and application of recommended dietary allowances, Nutr. Rev. 56, S2–S4.
- Lal R. (2009) Soils and food sufficiency. A review, Agron. Sustain. Dev. 29, 113–133.
- Lambert I.H., Hoffmann E.K., Pedersen S.F. (2008) Cell volume regulation: Physiology and pathophysiology, Acta Physiol. 194, 255–282.
- Lockett C.T., Calvert C.C., Grivetti L.E. (2000) Energy and micronutrient composition of dietary and medicinal wild plants consumed during drought. Study of rural Fulani, Northeastern Nigeria, Int. J. Food Sci. Nutr. 51, 195–208.
- Lopez M.V., Satti S.M.E. (1996) Calcium and potassiumenhanced growth and yield of tomato under sodium chloride stress, Plant Sci. 114, 19–27.
- Lopez-Berenguer C., Martinez-Ballesta M.C., Moreno D.A., Carvajal M., Garcia-Viguera C. (2009) Growing hardier crops for better health: Salinity tolerance and the nutritional value of broccoli. J. Agr. Food Chem. 57, 572–578.
- Losak T., Hlusek J., Kra S., Varga L. (2008) The effect of nitrogen and sulphur fertilization on yield and quality of kohlrabi (*Brassica oleracea*), R. Bras. Ci. Solo. 32, 697–703.
- Lukaski H.C. (2004) Vitamin and mineral status: Effects on physical performance, Nutrition 20, 632–644.
- Mäder P., Fliessbach A., Dubois D., Gunst L., Fried P., Niggli U. (2002) Soil fertility and biodiversity in organic farming, Science 296, 1694–1697.
- Mahouachi J. (2007) Growth and mineral nutrient content of developing fruit on banana plants (*Musa acuminata* AAA,'Grand Nain') subjected to water stress and recovery, J. Hortic. Sci. Biotech. 82, 839–844.
- Malmauret L., Parent-Massin D., Hardy J.L., Verger P. (2002) Contaminants in organic and conventional foodstuffs in France, Food Addit. Contam. 19, 524–532.
- Maqueda C., Ruiz J.C., Morillo E., Herencia J.F. (2001) Effect of an organic amendment on nutrient availability and plant content, 11th Int. Symp. Environmental Pollution and

its Impact on Life in the Mediterranean Region, MESAEP & SECOTOX, Proceedings Limasol (Chipre).

- Maynard D.N., Barker A.V., Minotti P.L., Peck N.H. (1976) Nitrate accumulation in vegetables, Adv. Agron. 28, 71–118.
- Mertz W. (1981) The scientific and practical importance of trace elements, Philos. T. Roy. Soc. B 294, 9–18.
- Moe S.M. (2008) Disorders involving calcium, phosphorus, and magnesium, Prim. Care, Clinics in Office Practice 35, 215–237.
- Montanaro G., Dichio B., Xiloyannis C., Celano G. (2006) Light influences transpiration and calcium accumulation in fruit of kiwifruit plants (*Actinidia deliciosa var. deliciosa*), Plant Sci. 170, 520–527.
- Moreno D.A., Villora G., Romero L. (2003) Variations in fruit micronutrient contents associated with fertilization of cucumber with macronutrients, Sci. Hortic. 97, 121–127.
- Morgan K.T. (2008) Nutritional determinants of bone health, J. Nutr. Elder. 27, 3–27.
- Narasinga Rao B.S. (2003) Anaemia and micronutrient deficiency, Natl. Med. J. India 16, 46–50.
- Neeser C., Savidov N., Driedger D. (2007) Production of hydroponically grown calcium fortified lettuce, Proceedings of the 1st International Symposium on Human Health Effects of Fruits and Vegetable, Acta Hort. 744, 317–321.
- Otero V., Barreal M.E., Merino A., Gallego P.P. (2007) Calcium fertilization in a kiwifruit orchard, Proceedings of the 6th international symposium on kiwifruit, Vols 1 and 2, Acta Hort. 753, 515–520.
- Pele L.C., Thoree V., Mustafa F., He S., Tsaprouni L., Punchard N.A., Thompson R.P.H., Evans S.M., Powell J.J. (2007) Low dietary calcium levels modulate mucosal caspase expression and increase disease activity in mice with dextran sulfate sodium-induced colitis, J. Nutr. 137, 2475–2480.
- Peters U., Chatterjee N., McGlynn K.A., Schoen R.E., Church T.R., Bresalier R.S., Gaudet M.M., Flood A., Schatzkin A., Hayes R.B. (2004) Calcium intake and colorectal adenoma in a US colorectal cancer early detection program, Am. J. Clin. Nutr. 80, 1358–1365.
- Ramirez-Serrano R., Larrinaga-Mayoral J.A., Murillo-Amador B., Hernandez-Saavedra N.Y., Fujiyama H. (2008) Antioxidant enzymatic response of hot pepper (*Capsicum annuum* L.) under saline stress conditions, Interciencia 33, 377–383.
- Rayssiguier Y., Gueux E., Nowacki W., Rock E., Mazur A. (2006) High fructose consumption combined with low dietary magnesium intake may increase the incidence of the metabolic syndrome by inducing inflammation, Magnesium Res. 19, 237–243.
- Recommended dietary allowance, Year of estimate, 2009, Available in: http://www.anyvitamins.com/rda.htm.
- Renkema K.Y., Alexander R.T., Bindels R.J., Hoenderop J.G. (2008) Calcium and phosphate homeostasis: Concerted interplay of new regulators, Ann. Med. 40, 82–91.
- Ríos J.J., Blasco B., Cervilla L.M., Rubio-Wilhelmi M.M., Ruiz J.M., Romero L. (2008) Regulation of sulphur assimilation in lettuce plants in the presence of selenium, Plant Growth Regul. 56, 43–51.
- Roe E. (1998) Compost utilization for vegetable and fruit crops, Hortscience 33, 934–937.
- Rosenthal J.J.C., Gilly W.F. (2003) Identified ion channels in the squid nervous system, Neurosignals 12, 126–141.

- Rouphael Y., Cardarelli M.T., Colla G. (2008) Yield, mineral composition, water relations and water use efficiency of grafted mini-watermelon plants under deficit irrigation, Hortscience 43, 730–736.
- Roychowdhury T., Tokunaga H., Ando M. (2003) Survey of arsenic and other heavy metals in food composites and drinking water and estimation of dietary intake by the villagers from an arsenic-affected area of West Bengal, India, Sci. Total Environ. 308, 15–35.
- Santamaria P., Elia A., Serio F., Todaro E. (1999) A survey of nitrate and oxalate content in retail fresh vegetables, J. Sci. Food Agr. 79, 1882–1888.
- Saure M.C. (2001) Blossom-end rot of tomato (*Lycopersicon* esculentum Mill) – a calcium – or a stress-related disorder? Sci. Hortic. 90, 193–208.
- Schwartz G. (2005) Molybdenum cofactor biosynthesis and deficiency, Cell. Mol. Life Sci. 62, 2792–2810.
- Schwarz J.R., Bauer C.K. (2004) Functions of erg K+ channels in excitable cells, J. Cell. Mol. Med. 8, 22–30.
- Sharpley A.N., Meisinger J.J., Power J.F., Suarez D.L. (1992) Root extraction of nutrients associated with long-term soil management, Adv. Soil Sci. 19, 151–217.
- Shenkin A. (2008) Basics in clinical nutrition: Physiological function and deficiency states of trace elements, e-SPEN 3, 255–258.
- Smith D.M., McKenna K., Thompson C.J. (2000) Hyponatraemia, Clin. Endocrinol. 52, 667–678.
- Sobotka L., Allison S., Stanga Z. (2008) Basics in clinical nutrition: Water and electrolytes in health and disease, e-SPEN 3, 259–266.
- Soupart A., Decaux G. (1996) Therapeutic recommendations for management of severe hyponatremia: Current concepts on pathogenesis and prevention of neurologic complications, Clin. Nephrol. 46, 149–169.
- Sritharan R, Lenz F. (1992) Effect of light regime on growth, carbohydrates and nitrate concentration in kohlrabi (Brassica oleracea var. gongylodes L.), J. Appl. Bot. 66, 130–134.
- Starck Z., Siwiec A., Chotuj D. (1994) Distribution of calcium in tomato plants in response to heat stress and plant growth regulators, Plant Soil 167, 143–148.
- Staugaitis G., Viskelis P., Venskutonis P.R. (2008) Optimization of application of nitrogen fertilizers to increase the yield and improve the quality of Chinese cabbage heads, Acta Agr. Scand. B-S. P. 58, 176–181.
- Steingröver E., Ratering P., Siesling J. (1986) Daily changes in uptake, reduction and storage of nitrate in spinach grown at low light intensity, Physiol. Plant. 66, 550–556.
- Szefer P., Grembecka M. (2007) Mineral components in food crops, beverages, luxury food, species, and dietary food, in: Szefer P., Nriagu J.O. (Eds.), Chemical and functional properties of food components series, CRC Press, Taylor and Francis Group, New York, USA, pp. 231–322.
- Theobald H. (2005) Dietary calcium and health, Nutr. Bull. 30, 237–277.
- Thomas V.L.K., Gropper S.S. (1996) Effect of chromium nicotinic acid supplementation on selected cardiovascular disease risk factors, Biol. Trace Elem. Res. 55, 297–305.
- Togay Y., Togay N., Erman M., Dogan Y. (2008) Performance of dry bean (*Phaseolus vulgaris*) as influenced by phosphorus and sulphur fertilization, Indian J. Agr. Sci. 78, 299–303.

- Tomala K., Dilley D.R. (1990) Some factors influencing the calcium level in apple fruits, Acta Hort. 274, 481–487.
- Tromp J. (1975) The effect of temperature on growth and mineral nutrition of fruits of apple, with special reference to calcium, Physiol. Plant. 33, 87–93.
- Turan M., Kordali S., Zengin H., Dursun A., Sezen Y. (2003) Macro and micro mineral content of some wild edible leaves consumed in eastern anatolia, Acta Agr. Scand. B 53, 129–137.
- Turner N.C. (2001) Optimizing water use, in: Nösberger J., Geiger H.H., Struik P.C. (Eds.), Crop science, progress and prospects, Third International Crop Science Congress, Hambourg, CABI, Wallingford, UK, pp. 119–135.
- Unal G., Akalin A.S., Akbulut N. (2007) Importance of dairy products in metabolic syndrome-obesity and hypercholesterolemia - Part 1, Agro. Food Ind. Hi. Tec. 18, 26–28.
- Val J., Monge E., Risco D., Blanco A. (2008) Effect of preharvest calcium sprays on calcium concentrations in the skin and flesh of apples, J. Plant Nutr. 31, 1889–1905.
- Vidal-Martinez J.L., Nunez-Escobar R., Lazcano-Ferrat I., Etchevers-Barra J.D., Carrillo-Gonzalez R. (2006) Potassium nutrition of broccoli (Brassica olearacea) with conventional management and fertigation in a vertisol in greenhouse, Agrociencia 40, 1–11.
- Víllora G., Moreno D.A., Pulgar G., Romero L. (1999) Zucchini growth, yield, and fruit quality in response to sodium chloride stress, J. Plant Nutr. 22, 855–861.
- Víllora G., Moreno D.A., Pulgar G., Romero L. (2000) Yield improvement in zucchini under salt stress: determining micronutrient balance, Sci. Hortic. 86, 175–183.
- Vogtmann H., Matthies K., Kehres B., Meier-Ploeger A. (1993) Enhanced food quality: Effects of compost on the quality of plant foods, Compost Sci. Util. 1, 82–100.
- Wang Z., Li S. (2004) Effects of N and P fertilization on plant growth and nitrate accumulation in vegetables, J. Plant. Nutr. 27, 539–556.
- Wang Z., Li S., Malhi S. (2008) Effects of fertilization and other agronomic measures on nutritional quality of crops, J. Sci. Food Agr. 88, 7–23.
- Wang Z., Tian X., Li S. (2001) The cause of nitrate accumulation in leafy vegetables, Acta Ecol. Sin. 21, 1136–1141.
- Warman P.R. (2005) Soil fertility, yield an nutrient contents of vegetable crops after 12 years of compost or fertilizer amendments, Biol. Agric. Hortic. 23, 85–96.
- Warman P.R., Havard K.A. (1997) Yield, vitamin and mineral contents of organically and conventionally grown carrots and cabbage, Agr. Ecosyst. Environ. 61, 155–162.

- Warman P.R., Havard K.A. (1998) Yield, vitamin and mineral contents of organically and conventionally grown potatoes and sweet corn, Agr. Ecosyst. Environ. 68, 207–216.
- Welch R.M., Graham R.D. (2004) Breeding for micronutrients in staple food crops from a human nutrition perspective, J. Exp. Bot. 55, 353–364.
- Welch R.M., Combs G.F. Jr., Duxbury J.M. (1997) Toward a 'green' revolution, Issues Sci. Technol. 14, 50–58.
- Williams C.M. (2002) Nutritional quality of organic food: Shades of grey or shades of green? Proc. Nutr. Soc. 61, 19– 24.
- Williams F.H. (2008) Neuromuscular complications of nutritional deficiencies, Phys. Med. Rehabil. Clin. N. Am. 19, 125–148.
- Winter C.K., Davis S.F. (2006) Organic foods, J. Food Sci. 71, R117–R124.
- Wojcik P., Wojcik M. (2003) Effects of boron fertilization on 'Conference' pear tree vigor, nutrition, and fruit yield and storability, Plant Soil 256 413–421.
- Wopereis M.C.S., Tamélokpo A., Ezui K., Gnakpénou D., Fofana B., Breman H. (2006) Mineral fertilizer management of maize on farmer fields differing in organic inputs in the West African savanna, Field Crop. Res. 96, 355–362.
- Wszelaki A.L., Delwiche J.F., Walker S.D., Liggett R.E., Scheerens J.C., Kleinhenz M.D. (2005) Sensory quality and mineral and glycoalkaloid concentrations in organically and conventionally grown redskin potatoes (*Solanum tubero-sum*), J. Sci. Food Agr. 85, 720–726.
- Xiao-Yun S., Guo-Zhen D., Hong L. (2006) Studies of a naturally occurring molybdenum-induced copper deficiency in the yak, Vet. J. 171, 352–357.
- Zhang F.C., Kang S.Z., Li F.S., Zhang J.H. (2007) Growth and major nutrient concentrations in *Brassica campestris* supplied with different NH<sup>+</sup><sub>4</sub>/NO<sup>-</sup><sub>3</sub> Ratios, J. Integr. Plant Biol. 49, 455–462.
- Zhou Y., Ren S., Wang Z. (1989) The effect of application of N fertilizer on the accumulation of nitrate in vegetables, Acta Agr. Boreali-Sin. 4, 110–115.
- Zhou Z.Y., Wang M.J., Wang J.S. (2000) Nitrate and nitrite contamination in vegetables in China, Food Rev. Int. 16, 61–76.
- Zimmermann M.B., Köhrle J. (2002) The impact of iron and selenium deficiencies on iodine and thyroid metabolism: Biochemistry and relevance to public health, Thyroid 12, 867–878.
# Fertiliser Trees for Sustainable Food Security in the Maize-Based Production Systems of East and Southern Africa

Festus K. Akinnifesi, O.C. Ajayi, G. Sileshi, P.W. Chirwa, and Jonas Chianu

#### Contents

1	Intro	duction	130
2	Ferti	liser Tree Systems	131
	2.1	The Faidherbia Albida System	131
	2.2	Sequential Tree Fallow	131
	2.3	Annual Relay Intercropping	131
	2.4	Gliricidia Intercropping	132
	2.5	Biomass Transfer	132
3	Ratio	onale for Promoting Fertiliser Trees	132
	3.1	Improvement in Crop Yield	132
	3.2	Soil Health	134
	3.3	Improvement in Soil Chemical Properties	135
4	Adoj	otion, Scaling Up and Impact	139
	4.1	Adoption	139
	4.2	Biophysical and Socioeconomic	
		Considerations for Proper Targeting	142
	4.3	Impact of Fertiliser Trees on Livelihoods	143
5	Conc	lusion	143
Re	eferenc	ces	144

F.K. Akinnifesi (🖂)

Abstract The negative effects of soil fertility depletion on food security, especially among smallholder farmers in Africa, is of economic importance, and may be worsened by climate change and rising global fertiliser prices. Substantial efforts and investment have gone into development of alternative soil fertility management options. These include vigorous research and development of N-fixing plants or "fertiliser trees", that has been on-going in the last two decades in East and Southern Africa. In this paper, we review several studies conducted both on-station and on-farm and synthesise the results in terms of improvements in soil physical, chemical and biological properties, and crop yield in response to fertiliser trees. Our major findings are that (1) fertiliser trees add more than  $60 \text{ kg N} \text{ ha}^{-1}$  per year through biological nitrogen fixation (BNF); (2) nutrient contributions from fertiliser tree biomass can reduce the requirement for mineral N fertiliser by 75%, translating to huge savings on mineral fertilisers; (3) fertiliser trees were also shown to substantially increase crop vield. A meta-analysis has further provided conclusive evidence that with good management, fertiliser trees can double maize yields compared with local farmer practices of maize cultivation without addition of external fertilisation. (4) Financial analyses showed that fertiliser tree systems are profitable and also have higher net returns than the farmers' de facto practice, i.e. continuous maize cropping without fertiliser. We conclude that widespread adoption and scaling up of fertiliser trees can reduce the amount of mineral fertiliser needed, maintain the soil ecosystem, and positively impact on the livelihoods of farm households in southern Africa.

World Agroforestry Centre (ICRAF), Southern Africa Regional Programme, PO Box 30798, Lilongwe, Malawi e-mail: fakinnifesi@africa-online.net

<sup>E. Lichtfouse et al. (eds.), Sustainable Agriculture Volume 2, DOI 10.1007/978-94-007-0394-0\_9,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Akinnifesi et al., Agron. Sustain. Dev. 30 (2010) 615–629. DOI: 10.1051/agro:2009058</sup> 

Keywords Agroforestry • Integrated soil fertility management•N-fixing legumes•Scaling up

#### 1 Introduction

Sub-Saharan Africa is often described as food-insecure (Conway and Toenniessen, 2003). The 2007-2008 food price crisis has not only increased food insecurity around the globe, but also exposed long-term failures in the functioning of the world food system (von Braun, 2009). Food shortage has nearly reached challenging dimensions and may become more crippling in the near future than anything the world has ever seen, unless efforts are geared toward improving productivity. At the centre of the imminent food catastrophe is maize, one of the main inputs in biofuel production as well as a staple food in most parts of Africa. Driven by a rapid rise in petroleum prices and, in response, a massive global expansion of biofuel production from maize (Cassman, 2007), the price of maize rose by over 50% from 2001–2007 (FAO, 2008). The rise in maize price not only impacts on the price of food products made from grains, but also the price of meat due to increased prices of livestock feed. Both food crops and feed demand are estimated to double in the next half century (Gowing and Palmer, 2008). The trends in southern Africa are worse than those in other parts of sub-Saharan Africa as population growth, dietary change and land-use pressures have driven prices of food and agricultural inputs to new heights. The prospects for meeting food demand in sub-Saharan Africa, which depends mainly on rain-fed and smallholder agriculture (Conway and Toenniessen, 2003), will likely remain bleak without major efforts to reverse current trends.

The focus of this work is the maize-based mixed farming system, which is the most important food production system in East and Southern Africa. It extends across plateau and highland areas at altitudes of 800–1500 metres, from Kenya and Tanzania to Zambia, Malawi, Zimbabwe, South Africa, Swaziland and Lesotho (Dixon et al., 2001). Maize accounts for 60% of the cropped area in some countries such as Malawi, Zimbabwe and Zambia, and it is almost a dominant crop in other countries including Kenya and Tanzania. In Malawi, maize is estimated to be grown on over 70% of the arable land and nearly 90% of the cereal area, making Malawi the world's highest consumer of maize at 148 kg per capita per year (Smale and Jayne, 2003). Thus, maize will remain a central crop in the food security equation even if the agricultural economy is diversified (Sauer et al., 2007). Crop-livestock integration is strong in the maize mixed farming system, where cattle are the most important livestock species. This farming system accounts for 10% of the land area and 19% of the cultivated area (Dixon et al., 2001). The climate varies from dry sub-humid to moist sub-humid. The most typical areas have unimodal rainfall, but some areas experience bimodal rainfall. The maize mixed farming system is currently in crisis (Dixon et al., 2001). Average farm sizes have fallen to under 0.5 ha in several areas (Dixon et al., 2001), while opportunities for expansion of cultivated land are limited as rapid population growth has led to progressive encroachment upon marginal lands (Bojo, 1996). Most farmers in the maize-based farming systems are crowded out of the agricultural input market and can hardly afford optimal quantities of inorganic fertiliser (Sauer et al., 2007).

The rapid deterioration of soils in this farming system directly affects productivity and it perpetuates rural poverty. Malawi alone loses US\$350 million worth of nitrogen and phosphorus through erosion each year, which translates to a gross annual loss of income equivalent to 3% of the agricultural Gross Domestic Product of Malawi (Bojo, 1996). If the situation is to be improved, agricultural production needs to be intensified through the application of agro-ecological technologies that do not require large amounts of capital and labour; a development paradigm termed the "Doubly Green Revolution" (Conway and Toenniessen, 2003). The fertiliser tree system is one of such innovations. A range of fertiliser tree options have been developed and several publications have documented individual studies (Kwesiga et al., 2003; Akinnifesi et al., 2008). There is need for an updated evidence-based review on the lessons learnt from about two decades of Research for Development on fertiliser tree technologies in terms of the science and their adoption and impact. Therefore, the objective of this review is to synthesise experiences in the development, scaling up and impact of fertiliser tree systems in the last two decades in southern Africa.

#### 2 Fertiliser Tree Systems

Fertiliser tree systems involve soil fertility replenishment through on-farm management of nitrogenfixing trees (Mafongoya et al., 2006). They represent a new paradigm because they use a completely different approach to land-use management by smallholder farmers. First, fertiliser tree systems capitalise on biological N fixation by legumes to capture atmospheric N and make it available to crops. Secondly, they permit growing of trees in association with crops in space or time to benefit from complementarity in resource use (Gathumbi et al., 2002). Thirdly, they address most of the biophysical and socioeconomic limitations identified with the earlier technologies based on using N-fixing tree legumes such as green manures (Kwesiga et al., 2003; Akinnifesi et al., 2006, 2008). The different fertiliser tree systems that have been developed and promoted in East and Southern Africa in the last two decades are briefly discussed below.

#### 2.1 The Faidherbia Albida System

The potential of faidherbia (Faidherbia albida) for improvement of soil fertility and crop yields has been demonstrated in many parts of Africa (Saka et al., 1994; Kang and Akinnifesi, 2000). This species has a unique phenology in that it sheds its leaves during the wet season and resumes leaf growth during the dry season. This makes it possible to grow crops under its canopy with minimum shading on the companion crop. About 20 to 30 mature trees are needed to completely cover one hectare of land and maintain optimum crop response (Kang and Akinnifesi, 2000). Several studies in Africa showed yield benefits when crops were grown under the canopy of Faidherbia. Saka et al. (1994) reported 100-400% increase in maize yield in the Lakeshore plain o Malawi.

However, it takes a long waiting period (up to 20 years) for the tree to reach maturity and have an impact on the under-storey crop (Kang and Akinnifesi, 2000). Recent development has shown that with closer spacing,  $10 \times 10$  m, earlier impact can be achieved at 12–15 years (Dutch Gibson, pers. comm.). A major improvement of this system is integration with

other sustainable land management options, such as use of short rotation fallow species in the first 10–15 years.

#### 2.2 Sequential Tree Fallow

Sequential tree fallow, often known as 'improved fallow', is a practice whereby a piece of land is planted with fast-growing nitrogen-fixing trees or shrubs for 2–3 years' fallow (Mafongoya et al., 2006). Tree fallows have distinct advantage over herbaceous fallows, particularly in seasonally dry climates, because they have the ability to tap nutrients from deeper soil layers and are capable of accumulating large quantities of biomass through which nutrients are recycled back for crop use. Nitrogen-fixing trees also add large quantities of N through biological nitrogen fixation and improve crop yield.

Improved fallows have been widely tested on farmers' fields in Zambia and this technology has now spread to other parts of southern Africa (Kwesiga et al., 2003). Several studies reviewed by Akinnifesi et al., (2008) showed that planted fallows of sesbania (*Sesbania sesban*) in Zambia, Malawi and Zimbabwe had doubled or tripled maize yield compared with control plots.

## 2.3 Annual Relay Intercropping

In relay intercropping, fast-growing nitrogen-fixing legumes are planted in a crop field at a time when annual crops such as maize have already been well established, usually within 2-4 weeks of crop sowing (Phiri et al., 1999). The legumes continue to grow after the crop harvest throughout the off-season. Legumes such as sesbania, tephrosia and pigeon pea (Cajanus cajan) are recommended. As farmers prepare land for the next season, they clear-cut the legume and incorporate the biomass into the soil. Although the yield levels are usually less than those of intercropping and 2-year improved fallow systems, it works well on small farms, and the benefit of trees can be seen immediately after one season of tree growth. Additionally, farmers do not lose any cropping year of maize. The main limitation of this technology is that the legumes need to be replanted every year.

#### 2.4 Gliricidia Intercropping

The intercropping of gliricidia (gliricidia sepium) with crops is an improvement building on the characteristics and advantages of alley cropping but minimising its biophysical limitations such as the "hedge effect", "competition" and tree management (Akinnifesi et al., 2006). A detailed description of this innovation has been published elsewhere (Akinnifesi et al., 2008). Gliricidia-maize intercropping has formed an important part of on-station and on-farm research in Malawi since the early 1990s (Akinnifesi et al., 2006). The socioeconomic and biophysical conditions in southern Malawi seem to meet most of the broadly defined criteria for the success of intercropping of crops with trees (Akinnifesi et al., 2006). The fact that land is scarce, labour is relatively cheap, fertiliser is costly in Malawi and the country is highly nitrogen-deficient, coupled with the fact that maize is a high nitrogen-demander, creates the prospect for adoption of gliricidia-maize intercropping in southern Malawi.

The main advantage of gliricidia intercropping is that once established it can be managed to continuously supply nutrients to crops year in, year out. Although gliricidia requires labour to establish seedlings and tree management, this is not yet a bottleneck as land holdings are less than a hectare and less than a quarter of a hectare is put to gliricidiamaize intercropping in the southern region of Malawi. Additionally, labour is cheap in Malawi due to high population density. Farmers appreciate that coppicing trees need to be established only once and can then be used for many years, despite low initial returns.

## 2.5 Biomass Transfer

Biomass transfer is essentially moving green leaves and twigs of fertiliser trees or shrubs from one location to another, usually in the wetlands to be used as green manure. Recent studies (Kuntashula et al., 2004) have shown that biomass transfer using fertiliser tree species is a more sustainable means for maintaining nutrient balances in maize and vegetable-based production systems. The advantage is that synchrony between nutrient release and crop uptake can be achieved with welltimed biomass transfer. The management factors that can be manipulated to achieve this are litter quality, rate of litter application, and method and time of litter application (Mafongoya et al., 1998).

Although it has been argued that biomass transfer technologies require a lot of labour for managing and incorporating biomass, economic analyses have concluded that it is unprofitable to invest in biomass transfer when labour is scarce and its cost is thus high (Kuntashula et al., 2004, 2006). In addition to increasing yields of vegetables such as cabbage, rape, onion and tomato, and maize grown after vegetable harvests, biomass transfer has shown potential to increase yields of other high-value crops such as garlic (Kuntashula et al., 2004, 2006).

# 3 Rationale for Promoting Fertiliser Trees

What is the evidence base for promoting fertiliser tree systems? The benefits of fertiliser tree adoption include significant increase in crop yield, improvement in soil health, and savings on mineral fertiliser costs and labour. Significant benefits are also derived from fertiliser trees in terms of other ecosystem services, including provision of fuelwood and fodder, reduction of erosion and carbon sequestration. The state of knowledge on the various ecosystem services of agroforestry has been reviewed by Sileshi et al., 2008. In the following sections we will briefly describe improvement in crop yields and soil health.

#### 3.1 Improvement in Crop Yield

One of the direct benefits of fertiliser trees is the maize yield response, as discussed in the next section. In an effort to fill the long-standing knowledge gap and conundrum of "marginal versus high impact" arguments regarding the effect of fertiliser trees on crop yield, a meta-analysis was undertaken using 94 peer-reviewed publications across sub-Saharan Africa (Sileshi et al., 2008). The results of the analysis provided a solid perspective for making recommendations about fertiliser trees to policy-makers, investors and scientists. Table 1 presents maize yields achieved using fertiliser trees across a range of sites in Malawi, Zambia, Zimbabwe

Species	Country	Number of sites	Yield (t ha <sup>-1</sup> )	Yield increase (t ha <sup><math>-1</math></sup> )	Percentage increase
Gliricidia	Malawi	5	3.9	2.9	345.6
	Tanzania	2	2.3	0.8	55.8
	Zambia	4	2.8	1.8	349.7
Sesbania	Malawi	7	2.5	1.3	161.4
	Tanzania	2	1.2	0.7	171.4
	Zambia	9	3.2	2.2	480.0
	Zimbabwe	4	3.0	1.9	583.1
Tephrosia	Malawi	9	2.0	1.1	232.7
	Tanzania	2	2.0	0.9	80.1
	Zambia	8	1.7	0.8	198.4
	Zimbabwe	5	3.6	0.2	17.7

**Table 1** Average maize yield and yield increase  $(t ha^{-1})$  with fertiliser trees relative to the control (unfertilised maize grown continuously) in southern Africa

Note: yield increase is the yield difference between the treatment (T) plot and the unfertilised control (C) plot, which is farmers' de facto practice. Percentage increase (%I) was calculated as follows: %I = 100((T-C)/C).



**Fig. 1** Long-term maize grain yield as affected by fertiliser and pruning incorporations in a gliricidia-maize intercropping in Makoka, Malawi. Arrows indicate flood due to excessive rainfall

in 1996/97, and droughts in the 1999/00 and 2003/04 seasons (Akinnifesi F.K., unpublished). Gs = *Gliricidia sepium*; N = nitrogen, P = phosphorus

and Tanzania. On average, gliricidia gave 55–350% yield increase over the control, while sesbania gave 160–583% increase. Yield increases with tephrosia spp were modest, and ranged from 180% to 233% (Table 1).

In a long-term trial in Makoka, gliricidia intercropping with maize increased maize yield in the range of 100 to 500%, averaging 315% over a ten-year period (Akinnifesi et al., 2006). Increase in yield is more apparent from the third year after tree establishment and onwards (Akinnifesi et al., 2006). The unfertilised plots not amended with gliricidia had steadily declining yield, and amendment with N and P could not sustain high maize yield over time (Fig. 1). Continuously cropped maize plots without gliricidia or fertiliser declined steadily from 2 t ha<sup>-1</sup> at the start of the experiment in 1992 to half a tonne in 2006. Unfertilised maize under gliricidia maintained yield at 3 to 4 t ha<sup>-1</sup>. When the intercrop plots were amended with 46 kg N ha<sup>-1</sup> and 40 kg P<sub>2</sub>O<sub>5</sub> ha<sup>-1</sup> (representing 50% N and 100% P, respectively), there was a 79% increase in grain yield over the recommended practice, indicating complementarity between the applied fertiliser and organic inputs from gliricidia (Akinnifesi et al., 2007). Similarly, in an on-farm experiment, 30% of the 40 on-farm type II farmers (farmer-managed trials) experienced increase in yield in the first two years, and 90% of these experienced yield increases in the subsequent two years (Akinnifesi et al., 2008). Yield increases in the third and fourth years averaged 69%. The authors observed that farmers with low yields in these early years were associated with poor field management conditions. Similarly, Makumba and Maghembe, (1999) reported yield increase in Makoka of 126% over three years for type I farmers' fields (researcher-managed), and an increase of 37% in on-farm type II averaged over five years. They attributed low response to erratic rainfall during the period.

## 3.2 Soil Health

Soil health has been broadly defined as the capacity of a living soil to function, within natural or managed ecosystem boundaries, to sustain biological productivity and diversity, maintain or enhance water and air quality, and promote plant and animal health (Doran, 2002; Sileshi et al., 2006a, b, 2008). In a global context, soil quality affects not only soil productivity but is also a significant factor governing environmental quality, and human and animal health and food safety and quality. Soil health is enhanced by management and land-use decisions that weigh the multiple functions of soil, and is impaired by decisions which focus only on single functions, such as crop productivity. Trees have been known to contribute to soil health in a number of ways: (i) enhancing soil physical structure and water regimes, (ii) improving soil chemical properties and nutrient input, (iii) increasing biological (microbial and faunal) communities, and (iv) suppressing soil pests. Several of these aspects of fertiliser tree management on soil health have been addressed in various studies in southern Africa. In the following section we will discuss these in detail.

#### 3.2.1 Improvement in Soil Physical Properties

Among the commonly used indicators of soil physical health are soil depth and rooting, infiltration, bulk density, water-holding capacity, aggregate stability, and penetration resistance. Fertiliser trees improve soil physical properties due to the addition of large quantities of litter fall, root biomass, root activity, biological activities, and roots leaving macropores in the soil following their decomposition (Rao et al., 1997). In studies conducted in eastern Zambia, sesbania fallows significantly increased the percentage of waterstable aggregates (>2 mm) compared with continuous maize cultivation without fertiliser (Sileshi and Mafongoya, 2006a). In the same experiment after two years of cropping, significantly lower bulk density and higher porosity (P < 0.05) was recorded in pigeon pea and sesbania fallows than a monoculture maize (Fig. 3). Similarly, bulk density was higher under monoculture maize compared with maize grown in association with gliricidia and L. leucocephala (Sileshi and Mafongoya, 2006a). The fact that fertiliser trees consistently improve soil physical properties is seen from measured increases in infiltration rates (Fig. 2), soil penetration resistance (Fig. 3), and reduced runoff and soil losses (Nyamadzawo et al., 2007; Phiri et al., 2003). Treatments involving fertiliser trees (leucaena, gliricidia, sesbania) have consistently shown significantly higher infiltration rates than monoculture maize (Fig. 2). Increased water infiltration implies reduced water runoff and thus low soil erosion. Generally, plots under fertiliser trees had lower resistance compared with continuously cropped maize plots (Chirwa et al., 2003; Fig. 3).

In a study conducted in Kagoro in eastern Zambia, the soil in maize planted following improved fallows had lower penetration resistance compared with monoculture maize at all soil depths (Fig. 3). The lower values following planted fallows could be attributed to the high amounts of litter biomass left on the surface by the fallow species. The lower infiltration and high penetrometer resistance in the monoculture maize indicate soil compaction as a result of degradation of soil structure. The improvement in soil structure under fertiliser trees was evident, as reflected by the results from timeto-runoff studies (Phiri et al., 2003). Rainfall simulation studies (Nyamadzawo et al., 2007) also indicated that sesbania and gliricidia mixed with Dolichos increased infiltration rates significantly compared with continuously fertilised maize plots.

In another study, Chirwa et al., 2007 reported that gliricidia did not compete with maize in a gliricidia-maize intercropping system in Makoka. The water-use efficiency (WUE) was higher in the agroforestry system than sole maize or pigeon pea.



**Fig. 2** Cumulative water intake in different treatments: (a) experiment 99-2, (b) experiment 2000-3, (c) experiment 92-2 in Masekera and (d) experiment in Kagoro (adapted from Sileshi and Mafongoya, 2006a and b; Chirwa et al., 2003)

## 3.3 Improvement in Soil Chemical Properties

Among the chemical indicators of soil health, total soil organic matter, the carbon to nitrogen ratio, carbon and nitrogen mineralisation rates, pH, electrical conductivity, and extractable N, P and K are commonly used. A recent study (Beedy, pers. comm., 2008) indicates that soil organic matter balance under gliricidia intercropping is positive following 14 years of continuous cropping. The study concluded that after 14 years, predictors of soil fertility and supporting soil organic matter fractions were significantly greater under the gliricidia-maize intercrop than under sole maize (Beedy, pers. comm., 2008). Both gliricidia intercrop and N fertiliser had a significant positive effect on dry season available N (Ikerra et al., 1999).

The legumes used in the sequential (e.g. fallow, relay) and simultaneous (e.g. intercrop) systems described above contribute to soil N through BNF and capture of subsoil N (otherwise unutilised by crops). Estimates of the amounts of N accumulated by fertiliser trees are given in Table 2. Out of the N accumulated, 55–84% is N derived from the atmosphere (Table 2). A series of multi-location trials were set up to measure the amount of N<sub>2</sub> fixed by different



**Fig. 3** Soil penetrometer resistance in different treatments: (a) in the 1999 cropping season in experiment 92-2 in Msekera (adapted from Sileshi and Mafongoya, 2006a) and (b) in Kagoro

(adapted from Chirwa et al., 2003). Treatments significantly differed in 1998 (P = 0.004) and 1999 (P = 0.018)

**Table 2** Amount of N fixed (kg/ha) and the percentage of nitrogen derived from the atmosphere (% Ndfa range) by fertiliser trees in southern Africa

Species	N fixed	%Ndfa	Site (Country)	References*
Acacia anguistissima	122	56–79	Chikwaka (Zimbabwe)	1
	210		Chipata (Zambia)	2
Pigeon pea	NA	65-84	Chikwaka (Zimbabwe)	1
	64	96–99	Nyambi (Malawi)	3
	85	94–97	Ntonda (Malawi	3
	34	66–96	Gairo (Tanzania)	3
	54	95–99	Babati (Tanzania)	3
Gliricidia sepium	212	NA	Chipata (Zambia)	2
Leucaena collinsii	300	NA	Chipata (Zambia)	2
Sesbania sesban	84	55-84	Chikwaka (Zimbabwe)	1
Tephrosia candida	280	NA	Chipata (Zambia)	2
Tephrosia vogelii	157	NA	Chipata (Zambia)	2

\* References: 1. Chikwko et al. (2004); 2. Mafongoya et al. (2006); 3. Adu-Gyamfi et al. (2007).

tree genera and provenances using the <sup>15</sup>N natural abundance method in Zambia. The data shows high variability among species and varieties of the same species in percent N derived from the atmosphere (Ndfa). So the measurement task is still a challenging one (Mafongoya et al., 2006). Two-year tree fallows of the non-coppicing species sesbania and tephrosia are able to replenish soil N to levels sufficient to grow three subsequent high-yielding maize crops in southern Africa (Kwesiga and Coe, 1994). Unlike non-coppicing species, coppicing trees such as gliri-

cidia and *Leucaena* spp. cause increases in residual soil fertility beyond 2–3 years because of the additional organic inputs that are derived each year from coppice re-growth that is cut and applied to the soil. The fertiliser value of total N was estimated to exceed  $60-75 \text{ kg N ha}^{-1}$  (Akinnifesi et al., 2008), which can replace the current need for mineral N. Some legumes were more effective in improving soil productivity and maize yield than others, probably due to differences in biomass production, N<sub>2</sub> fixation and recovery of leached nutrients.

Legumes can also have other beneficial effects on crop yield as they can improve availability and uptake of nutrients such as phosphorus. In small-scale farming systems in Africa, crop harvesting removes almost all of the P accumulated by cereal crops (Sanchez et al., 1997). Application of plant biomass from fertiliser trees as green manure can contribute to P availability, either directly by releasing tissue P during decomposition and mineralisation or indirectly by acting on chemical processes that regulate P adsorptiondesorption reactions (Mweta et al., 2007). Table 3 presents the P and K input from biomass for different fertiliser tree species. Soil organic matter contributes indirectly to raising P in soil solution by complexing certain ions such as Al and Fe that would otherwise constrain P availability (Li et al., 2003; Mweta et al., 2007). Decomposing organic matter also releases anions that can compete with P for fixation sites, thus reducing P adsorption. The more extensive root systems that trees and shrubs have compared to crops increase the exploration of larger soil volumes, which results in enhanced uptake of P and other nutrients (Schroth, 1999).

Rotation of maize with legume fallows can result in more effective subsoil nitrate and water utilisation than maize monoculture (Chirwa et al., 2007; Nyamadzawo et al., 2007; Phiri et al., 2003). Where both soil organic matter and phosphorus are very poor, legumes may not accumulate a significant amount of biomass and will fix little N. To maintain positive nutrient balances for N and P in these environments, organic resources need to be combined with low rates of mineral fertiliser amendment (Ajayi et al., 2009).

The retrieval and cycling of nutrients from soil below the zone exploited by crop roots is referred to as nutrient pumping. Deep capture is favoured when perennials have a deep rooting system and a high demand for nutrients, when water or nutrient stress occurs in the surface soils, and/or when extractable nutrients occur in the subsoil (Buresh and Tian, 1997). These conditions were observed in eastern Zambia where nitrate accumulated in the subsoil during periods of maize growth. Fertiliser trees grown in rotation with maize could effectively retrieve the nitrate in the subsoil that is not accessible to maize (Mafongoya et al., 2006). Intercropping rather than rotating fertiliser trees with crops appears to improve the long-term efficiency of nutrient use in deep soils. The nutrient balance has been shown to be positive after 8-12 years of continuous cultivation with fertiliser trees such as gliricidia in Malawi and Zambia (Akinnifesi et al., 2007; Mafongoya et al.,

		Nutrien	t input							
Tree species	Tree management	N	Р	K	Site	Reference				
G. sepium	Coppicing	33.7	2.0	21.4	Muheza (Tanzania)	1				
	Pollarding	71.9	4.4	45.8	Muheza (Tanzania)	1				
L. leucocephala	Coppicing	65.6*	3.6	30.9*	Msekera 1 (Zambia)	2				
	Coppicing	44.3 <sup>†</sup>	$2.5^{\dagger}$	$20.6^{\dagger}$	Msekera 2 (Zambia)	2				
G. sepium	Coppicing	69.9*	4.6*	26.2*	Msekera 1 (Zambia)	2				
	Coppicing	69.2 <sup>†</sup>	4.6†	$25.9^{\dagger}$	Msekera 2 (Zambia)	2				
	Coppicing	72.1	-	-	Kagoro (Zambia)	3				
	Coppicing	67.3	-	-	Kalunga (Zambia)	4				
	Coppicing	74.4	5.2	42.5	Makoka (Malawi)	5				
S. sesban	Non-coppicing	38.0	-	-	Chikwaka (Zimbabwe)	6				
Pigeon pea	Non-coppicing	82.0	-		Chikwaka (Zimbabwe)	6				
G. sepium	Coppicing	oppicing - 2.2 13.2 Msekera 3 (Zambia)		Msekera 3 (Zambia)	4					
	Coppicing	-	4.3	25.3	Kalunga (Zambia)	4				

**Table 3** Annual inputs of the major nutrient (kg/ha) from biomass<sup>‡</sup> from fertiliser trees added to the soil

<sup>‡</sup> In the case of coppicing species this represents only coppice biomass, while in non-coppicing species both litter and standing leaf biomass are considered.

\* Averaged over 9 years.

† Averaged over 5 years. Msekera 1 and 2 represent experiments 92-3 and 97-3, respectively.

<sup>††</sup> References: 1. Meliyo et al., (2007); 2. Sileshi and Mafongoya, (2006a); 3. Chirwa et al., (2003); 4. Sileshi and Mafongoya, (2006b); 5. Akinnifesi et al., (2006); 6. Chikowo et al., (2004).

2006). Intercropping with fertiliser trees such as gliricidia may be more effective for pumping of soil nutrients than a fallow legume-maize rotation. The introduction of gliricidia with maize rotation has a great potential for deep capture of Ca and Mg compared with continuously fertilised monoculture maize.

#### 3.3.1 Soil Biological Processes and Functions

Soil biological processes, mediated by roots, flora and fauna, are an integral part of the functioning of natural and managed ecosystems. Soil biota have been identified as potential indicators of soil health and sustainability at the farm level (Sanginga et al., 1992). These include microflora numbers, microbial biomass, enzyme activity and respiration, and soil fauna (abundance, diversity and community structure of soil arthropods, earthworms, etc.), as they respond sensitively to land management practices and correlate well with beneficial soil functions including water storage, decomposition and nutrient cycling, and suppression of pestiferous organisms.

Soil microflora such as fungi and bacteria are responsible for the breakdown of plant litter and most soil activities. Very few studies have examined the effect of fertiliser trees on soil biological properties. In a study conducted in Zimbabwe using leaf biomass of various fertiliser trees, microbial biomass carbon and nitrogen did not differ among treatments. However, fungal Actinomycetes populations differed with the biomass of legume species used as well as the method of biomass application (Mafongoya et al., 1997).

Among the macrofauna essential in soil processes in agro-ecosystems, probably the most important ones are the so-called ecosystem engineers (termites, earthworms and some ants), and the litter transformers including millipedes, some beetles and many other soil-dwelling invertebrates. Earthworms can be used as an integrative measure of soil health, assuming their importance in regulating soil processes which are vital to the continued formation of soil and as protection against soil degradation. These have been used to monitor changes in soil quality and to provide early warning of adverse trends and identify problem areas. In five separate experiments conducted in eastern Zambia, the number of invertebrate orders per sample and the total macrofauna (all individuals per square metre) recorded were higher when maize was grown

in association with tree legumes than under fertilised monoculture maize. Similarly, densities of earthworm and millipede were also higher than under monoculture maize (Sileshi and Mafongoya, 2006a, b). Cumulative litter fall, tree leaf biomass, and re-sprouted biomass under legume species appeared to explain the variation in macrofauna densities (Sileshi and Mafongoya, 2007). Litter transformer populations were higher under gliricidia, which produced good quality organic inputs, than among the other fallow species. On the other hand, a higher population of ecosystem engineers was found under trees that produce poor quality organic inputs (Sileshi and Mafongoya, 2006a, b).

#### 3.3.2 Reduction in Weed Problems

Declining soil fertility, along with the concomitant problems of weeds, pests and diseases is now a significant constraint to Africa's aspiration for sustainable development and food security (Sanchez, 2002). The declining soil resource base has also contributed to loss of biodiversity and persistent soil pest problems and weeds such as Striga spp. (witchweed). The effect of fertiliser trees on weeds and soil insects has been studied in eastern Zambia (Sileshi and Mafongoya, 2003; Sileshi et al., 2005, 2006). Abundance of Striga asiatica was significantly influenced by the quantity and the interaction effect of quantity and quality of biomass. Species that produce low to medium quantities of slowdecomposing biomass tended to reduce striga abundance in maize, while fast-decomposing ones did not (Sileshi et al., 2006). Similarly, in East Africa, reduction of another witchweed (Striga hermontica) by legume fallows depended on the rate of decomposition and nitrogen mineralisation of organic residues, which in turn was determined by quality in terms of carbon to nitrogen + polyphenol ratios (Gacheru and Rao, 2001). This indicates that the mechanism by which legume fallows influence striga is much more complicated than just by soil fertility improvement. Among the legumes tested, sesbania appeared to be the best in reducing striga infestation in maize in eastern Zambia (Sileshi et al., 2006).

Fertiliser trees have also reduced arable weed problems (Sileshi and Mafongoya, 2003; Sileshi et al., 2006). The mechanism by which legumes suppress arable weeds varies. Rotational fallows can modify the chemical ecology of the soil by releasing a range of volatile and water-soluble compounds that may act as germination stimulants or inhibitors. Chemicals such as nitrate and ethylene stimulate germination of numerous agricultural weeds. These compounds also sensitise weed seeds to other environmental factors such as changes in soil temperature and exposure of weed seeds to light (Sileshi et al., 2006).

#### 3.3.3 Reduction in Soil Insects

Although termites are generally essential ecosystem engineers, some are also crop pests. Few, if any effective methods exist to control pestiferous species. Fertiliser tree systems generally reduce insect pests such as termites (Sileshi and Mafongoya, 2003; Sileshi et al., 2005). In a study conducted in eastern Zambia, Sileshi and Mafongoya, (2003) recorded lower termite damage (% lodged plants) on maize planted after tephrosia + pigeon pea, sesbania + pigeon pea and pure sesbania compared with maize grown after traditional grass fallow. Monoculture maize grown after traditional grass fallow had about 11 and 5 times more termite damage compared with maize grown after tephrosia + pigeon pea and sesbania + pigeon pea, respectively. In another set of experiments, Sileshi et al., (2005) monitored termite damage on maize grown in coppicing fallows. Those studies showed that fully-fertilised monoculture maize suffered higher termite damage compared with maize grown in gliricidia and L. leucocephala.

## 4 Adoption, Scaling Up and Impact

Given the biophysical performance and relevance of fertiliser trees in southern Africa, since the mid-1990s, emphasis on the system has shifted from purely onstation field trials to on-farm research, which allows incorporation of socioeconomic studies of adoption, profitability, labour, farmer perception and acceptability of different fertiliser tree systems under farmers' field conditions (Ajayi, 2007). The research for development efforts has therefore been expanded to address questions on farmer uptake, determinants of adoption and factors influencing farmers' decisions to adopt fertiliser trees, impacts of the technological innovations, and constraints and obstacles against adoption.

#### 4.1 Adoption

A number of empirical studies have been carried out to gain insights into the factors influencing farmers' decisions to adopt fertiliser trees and the impacts that the technology has made on livelihoods and the environment in southern Africa. Using a logistic regression approach, Thangata and Alavalapati, (2003) investigated the adoption of mixed inter-cropping of Gliricidia sepium and maize in Malawi. Their results suggest that age of the farmer, frequency of contact with extension, and the effective number of household members who contribute to farm work are important variables determining the adoption of agroforestry. A study in Zambia (Keil et al., 2005) found that 75% of farmers who initially tested fertiliser trees eventually adopted the technology. Their study shows that scarcity of capital, inadequate access to markets for fertiliser and relatively low population density are the conditions that enhance the adoption of the technology. Studies on the use of labour in agricultural field plots in Zambia show that over a five-year period, the total quantity of labour used in fertiliser tree plots was 13% lower than unfertilised maize, and far less compared with fertilised monoculture maize plots (Franzel et al., 2002; Franzel, 2004). A study by Ajayi et al., (2007) found that aggregated over a five-year cycle, the total quantity of labour input used in fertiliser tree plots (improved fallows) was lower than in fertilised continuously cropped maize fields, but higher than in nonfertilised maize. These results do not lend credence to the notion that fertiliser trees are more labour-intensive given that the quantity of labour inputs used per unit of fertiliser tree plot area is not higher than in fertilised maize. Given the small plot sizes of fertiliser trees, estimated at an average of 0.2 ha only, farmers' decision to test fertiliser tree systems or not may not be attributed to the quantity of labour requirements. Rather, the popular perception regarding labour constraints and adoption of fertiliser trees in fertiliser trees may be due to the fact that some field operations may coincide with operations in other fields (especially cash crops) that are managed by the same households, and which depend on the same labour supply drawn from household members (Ajayi et al., 2009; Ajayi, 2007). This suggests that both the quantity and temporal distribution of labour input requirements are important factors in farmers' decision to adopt fertiliser trees. It is expected that as the land area that farmers cultivate

to agroforestry increases, the temporal distribution of labour requirement for tree establishment and management may become more significant than it is presently. A modification to the agronomic practices of the technology to shift some of the labour inputs away from the main cropping season to the "off peak" labour demand season is expected to enhance the acceptability of fertiliser trees among farmers. Based on these and several other studies, the main factors that affect the adoption of fertiliser trees have been identified. These can be grouped into four categories: household-specific, technology-specific, institution and policy, and geospatial factors (Ajayi et al., 2007), further elaborated below.

Household-specific factors: These include farmer perceptions, resource endowment, household size (a proxy for household labour supply), risk, and access to information on inputs and output prices. These factors vary widely across households, resulting in different levels of uptake of fertiliser trees by different typologies of farm households. Those households who have access to a larger pool of labour supply, e.g. higher household size or land and other production inputs tend to have higher levels of adoption (Ajayi et al., 2006; Keil et al., 2005). While economic performance and short-term profitability of fertiliser trees enhance the probability of farmers' uptake, these alone do not provide an exclusive explanation for farmers' adoption patterns. Key attitudinal issues such as farmers' perceived usefulness of the technology (Ajayi, 2007), and household resource endowment are important for adoption. Although most options of fertiliser trees have positive net present values over time, some of them attain break-even point only after two years, implying that farmers make an upfront investment for a couple of years before receiving returns to their investment in the technology. This poses challenges to some types of farm households in southern Africa, who may not be sufficiently well off to absorb the initial investment and/or who may want to derive immediate benefits from the technology (Ajayi et al., 2007). For some farmers, a long "waiting period" can forestall the adoption of certain fertiliser tree technologies that guarantee high net returns in future.

*Technology-specific factors*: The technology-specific factors that affect farmers' uptake of fertiliser trees include the management regime required under some

options as well as characteristics of particular fertiliser tree technology. Smallholder farmers more readily adopt specific options of fertiliser trees if such options produce grain that could be consumed or sold for cash income, in addition to replenishing their soils (Ajayi, 2007). Different types of fertiliser trees require varying amounts of labour and this plays an important role in their acceptability to farm households, depending on their internal labour endowment or ability to command additional labour from outside the household. In general, species that can be directly sown are much more preferred by farmers than those which require nursery establishment, transplanting, and other operations that add to the complexity of the options. Apart from the quantity of labour required to manage fertiliser trees, the temporal distribution of the same is also important for adoption (Ajayi et al., 2007). Fertiliser trees are an emerging technology relative to conventional agricultural practices that farmers have known, been used to, and have received training on for a much longer period. Unlike annual crop production technologies and conventional soil fertility management options, fertiliser tree systems require skills in terms of management of the trees.

In terms of profitability, fertiliser tree systems are profitable and have positive net benefits (Franzel et al., 2002; Franzel, 2004). A field study in Zambia (Ajayi et al., 2009) found that the net present value of maize plots amended with only fertiliser tree systems (US\$233–309) compared well with a full fertiliser dose (US\$349), and performed better than a continuous unfertilised maize plot; US\$130 (Table 4). In addition, the return to labour in fertiliser trees is two times higher than in unfertilised fields (Franzel, 2004). Improved fallows require 13% less labour inputs per hectare than unfertilised maize and 33% less labour inputs than fertilised maize (Franzel, 2004).

*Policy and institution factors*: The policy and institution context within which fertiliser trees are disseminated plays an important role in affecting decision-making regarding the technology. Such factors include input and output prices, customary land-use practices, land tenure and property rights. Policy and institutions are cross-cutting and affect several farmers because the adoption of a relatively longterm technology such as fertiliser trees depends on incentives created by market and non-market institutions (Ajayi et al., 2007). National policies may modify the profitability of fertiliser trees, thereby altering their

Description of system	Benefit-cost ratio	Net present value (US\$ /ha)	% increase in net profit over unfertilised maize
Continuous maize – non-fertilised	2.01	130	0
Continuous maize – subsidised fertiliser <sup>†</sup>	2.65	499	284
Continuous maize – fertiliser priced at market rate <sup>††</sup>	1.77	349	168
2-yr Gliricidia sepium fallow	2.91	269	107
2-yr sesbania fallow	3.13	309	138
2-yr tephrosia fallow	2.77	233	79

Table 4 Financial profitability of maize production systems using tree fallows, fertiliser and farmers' practices in Zambia<sup>†††</sup>

<sup>†</sup> Fertiliser subsidised by government at 50%.

<sup>††</sup> Fertiliser at market rates.

<sup>†††</sup> Figures are on a one hectare basis, at prevailing costs & prices and annual discount of 30%.

attractiveness and potential adoptability by farmers. Lack of access to quality seeds is one of the greatest constraints to fertiliser trees. Private sector organisations have not yet engaged in the multiplication and distribution of fertiliser tree seeds as done for the seeds of food crops such as maize. One of the reasons is that the market size and potential returns on investment in the latter is expected to be more rewarding for private entrepreneurs because more farmers currently grow maize than fertiliser trees. A profitability analysis conducted on fertiliser trees in Zambia showed that the four factors that most influenced the financial attractiveness and potential adoptability of the technology are external to the household, and most smallholder farm households have very little or no control over them (Ajayi et al., 2009).

Some local customary practices affect the nature of risk and potential adoptability of fertiliser trees. Field studies in Zambia show that bush fires and browsing constrain widespread adoption of certain fertiliser technologies (Ajayi and Kwesiga, 2003). Extensive browsing by livestock led to the discontinuation of the promotion of pigeon pea-based fertiliser trees in Zambia (Franzel et al., 2002). In addition, local customary practices and institutions (especially incidence of bush fires and browsing by livestock during the dry season, and absence of perennial private rights over land) prevailing in southern Africa limit widespread uptake of some agroforestry technologies (Ajayi and Kwesiga, 2003). Collaborative efforts initiated by traditional rulers, and research and development organisations to respond to these challenges have contributed to solving some of the constraints posed by these customary practices, e.g. through the enactment of bye-laws against the practices, but have not completely resolved them. Short-term customary land

tenure creates a disincentive to longer-term investment in tree-based technologies.

Geo-spatial factors: There is a spatial dimension to the adoption of fertiliser trees in southern Africa, as the performance of the technologies varies with location, across crops and with time. Geo-spatial factors focus on the performance of species across different biophysical conditions and site or village location. They include the type and characteristics of soils, which determine the bio-physical limits of technologies, access to roads and markets, and location of a village relative to institutions promoting fertiliser trees. The choice of species used for fertiliser trees is critical as the biophysical performance and social-economic needs of different communities vary from one region to another. The establishment of proper targeting of fertiliser trees to geographic and social niches is an important factor that affects the relevance of the technology to farmers and that they create the desired impact among smallholder farmers.

The fertiliser tree system is financially profitable, but its widespread uptake by smallholder farmers may be constrained by challenges posed at the farm, household and policy levels as enumerated above. One of the important lessons learnt is that scaling up of fertiliser trees requires both vertical processes (to influence policies and institutions that are conducive for farmer adoption) and horizontal processes (to quicken the spread of the technology across communities and geographic boundaries). In addition, there is the need for appropriate structures that support the uptake of fertiliser trees. Such structures include the existence of strategic partnerships with several research, education and development institutions, and viable seed and output markets.

# 4.2 Biophysical and Socioeconomic Considerations for Proper Targeting

After two decades of research, it is known that the technical performance of fertiliser trees is important but not an exclusively sufficient condition to guarantee their adoption by smallholder farmers. A substantially large volume of new knowledge has been generated on where these legume-based technologies fit best within spatially heterogeneous landscapes. The most important considerations include:

- Landholding size: In areas where landholding is a problem such as in the southern region of Malawi, Gliricidia is best suited as a permanent system where fallowing and cropping are concurrent. Improved fallows are not appropriate where landholding is small, such as southern Malawi where average holding is less than one hectare. The twoto three-year waiting period may also be a disincentive in land-pressured areas as farmers may be unable to allocate a separate field for fallows for such long periods. A well-designed simultaneous intercropping or relay fallow cropping system is ideal for such situations.
- 2. Waiting period: Where landholding size is bigger, many fertiliser trees and shrubs can be practised. For instance, faidherbia is recommended where a farmer can afford to wait for at least 12–15 years before getting soil fertility benefit. To reduce this period, we recommend that short-duration species be used as fertiliser during the first 10–12 years. Gliricidia and faidherbia are not recommended for farmers without permanent lands. On the other hand, farmers without permanent ownership of land, but access for a few years, would prefer short-duration species.
- 3. Land and tree tenures: Where the small landholding problem is also coupled with land tenure due to tree tenure, especially in matrilineal matrilocal systems, annual relay fallow intercropping with short-duration fallow species such as sesbania, *Tephrosia* spp. and pigeon pea become more attractive. In such matrilineal systems men are less motivated to plant permanent trees. Faidherbia and gliricidia are not recommended where land and tree tenures are a problem.
- 4. Nursery investment: *Tephrosia* spp. has the advantage that it could be sown directly without

nursery investment. Because gliricidia is planted once, farmers who are not interested in annual or bi-annual nursery establishment will prefer gliricidia. However, because of the shortage of seeds and need for nursery establishment, many farmers are encouraged to use *T*. candida. This also means that planting of gliricidia, and especially faidherbia, must be started early before the season, as these require 6–8 weeks for gliricidia and sesbania and 9–12 weeks for faidherbia before transplanting. Where the season has already begun, only pigeon pea or *Tephrosia* spp. are feasible as they could be sown in the field directly.

- 5. Germplasm availability: Availability of tree seeds may limit the type of fertiliser tree technology to embrace. Gliricidia is more expensive and difficult to obtain in large amounts. On the other hand, *Tephrosia* spp., faidherbia, pigeon pea and sesbania are prolific seed producers. However, gliricidia can be established from stem cuttings.
- 6. Soil type and catena positions: The fertiliser trees have specific niches they perform best. The survival of gliricidia on wetlands is generally poor. Well-drained soils are better for gliricidia. Sesbania is well suited to both well-drained and wetlands. However, it does not perform well on sandy soils. *Tephrosia* spp. performs well on flat to upper slopes, like gliricidia.
- 7. The results from the meta-analysis indicate that they are generally best-performing in low to medium potential sites in terms of rainfall and fertility.
- 8. Pests and disease: Some species (e.g. sesbania) are susceptible to pests during the seedling stage.
- 9. Grazing problem: where livestock grazing is a chronic problem, sesbania, pigeon pea and gliricidia may not be successful. In that case, *Tephrosia* spp. is ideal. Also, communities could formulate bye-laws to deal with animal encroachment during and after cropping seasons.
- 10. In some cases, the biomass produced by fallows could be constrained by low soil fertility and a supplement with micro-doses of inorganic fertilisers, especially P, is worthwhile. The use of P fertilisers from inorganic fertiliser or rock phosphate has been recommended for poor P-deficient soils (Sanchez, 2002).

# 4.3 Impact of Fertiliser Trees on Livelihoods

A large body of literature has been generated since the 1990s, and new results, innovations and challenges have emerged. Conclusions from these clearly indicate that agroforestry is making a positive impact on the livelihoods of people who adopt the technologies, although much remains to be done in quantifying the impact. One of the major impacts of fertiliser trees is on food security through the increase in maize yield (see Sect. 3.3). For example, an extra increase in yield equivalent to between 54 and 114 extra-person days of maize consumption reduced the hunger period by 2– 3 months per household in Zambia (Ajayi et al., 2007).

A monitoring and evaluation framework used with partners in five countries showed that the number of farmers benefiting from the technology increased from a few hundreds in the mid-1990s to more than 400000 by 2007 in the region (Schuller et al., 2005). An impact assessment in the region also indicated that farmers have generally increased the land under agroforestry, and appreciated that the fertiliser tree technologies have improved soil fertility for 59–84% of farmers, increased maize yield for 70–90% of practising farmers, improved food security and reduced hunger months by at least 2 months for 54–94% of farmers, and contributed to fuel wood availability for 54–90% of farmers, and other livelihood indicators (Table 5) as shown below.

Empirical studies in Zambia show that farmers appreciate fertiliser tree technology for its ability to respond to the critical problems of poor soil fertility, its effects on food security, and the additional benefits obtainable from fertiliser trees to households (Ajayi, 2007). They mentioned, however, challenges to the widespread uptake of the technology to include land constraints, land tenure rights, lack of tree seeds, and the knowledge-intensive nature of the technology.

## **5** Conclusion

From the discussion above it can be concluded that the fertiliser trees can sustain crop yield and deliver a range of other benefits that enable farmers to produce adequate food to feed their families. A variety of fertiliser tree systems have been developed to fit into different farming systems and farmers' socioeconomic circumstances. Although the technical performance of fertiliser trees is important, it is not exclusively sufficient to guarantee their adoption by smallholder farmers. Compared with technologies based on annual crops, the adoption of fertiliser trees will be slower because the technology involves multiple components and the multi-years through which testing, modification and uptake of the technology by farmers take place. Moving fertiliser trees to the mainstream requires approaches that overcome the major adoption hurdles discussed above: robust technology that fits farming systems, clear economic benefits from farmers' perspective, establishment of a sustainable germplasm base and input supply system, and a supportive macro-economic policy environment.

	Malawi	Zambia	Mozambique
Impact indicator	(n = 31)	(n = 184)	(n = 57)
Increase in area under Agroforestry	55	87	65
Yield increases (>quarter to tripled)	70	90	71
Significant food security (>2 months of hunger reduction)	94	84	54
Increase in income	58	68	53
Firewood availability	90	$\mathrm{nd}^\dagger$	59
Increased savings	87	94	71
Change in wealth	77	84	77
Strong reduction in Striga spp.	90	93	88
Soil improvement	84	82	59
Other benefits	$65^{\dagger\dagger}$	nd	24

**Table 5** Assessment of the impact of agroforestry adoption on livelihoods of farmers in Malawi, Mozambique and Zambia (Schueller et al., 2005)

<sup>†</sup> nd, not determined.

<sup>††</sup> Malawi, seed sale; Mozambique, tree stakes.

\* Figures in table represent % of respondents.

Acknowledgements The authors wish to acknowledge with gratitude the financial support for the Southern Africa Agroforestry Regional Programme by various investors including the Canadian International Development Agency (CIDA), Irish Aid, Ireland, DFID, European Commission, Rockefeller Foundation, Swedish International Development Agency (Sida), USAID and NORAD.

## References

- Adu-Gyamfi J.J., Myaka F.A., Sakala W.D., Odgaard R., Vesterager J.M., Høgh-Jensen H. (2007) Biological nitrogen fixation and nitrogen and phosphorus budgets in farmermanaged intercrops of maize–pigeonpea in semi-arid southern and eastern Africa, Plant Soil 295, 127–136.
- Ajayi O.C. (2007) User acceptability of soil fertility management technologies: lessons from farmers' knowledge, attitude and practices in southern Africa, J. Sustain. Agr. 30, 21–40.
- Ajayi O.C., Kwesiga F. (2003) Implications of local policies and institutions on the adoption of improved fallows in eastern Zambia, Agroforest. Syst. 59, 327–336.
- Ajayi O.C., Akinnifesi F.K., Sileshi G., Chakeredza S. (2007) Adoption of renewable soil fertility replenishment technologies in southern African region: lessons learnt and the way forward, Nat. Res. Forum 31, 306–317.
- Ajayi O.C., Akinnifesi F.K., Sileshi G., Kanjipite W. (2009) Labour inputs and financial profitability of conventional and agroforestry-based soil fertility management practices in Zambia, Agrekon 48, 276–292.
- Ajayi O.C., Massi C., Katanga R., Kabwe G. (2006) Typology and characteristics of farmers planting improved fallows in southern Africa, Zambian J. Agr. Sci. 8, 1–5.
- Akinnifesi F.K., Chirwa P.W., Ajayi O.C., Sileshi G., Matakala P., Kwesiga F.R., Harawa H., Makumba W. (2008) Contributions of agroforestry research to livelihood of smallholder farmers in Southern Africa: 1. Taking stock of the adaptation, adoption and impact of fertilizer tree options, Agric. J. 3, 58–75.
- Akinnifesi F.K., Makumba W., Kwesiga F. (2006) Sustainable Maize Production using Gliricidia/maize intercropping in Southern Malawi, Exp. Agr. 42, 441–457.
- Akinnifesi F.K., Makumba W., Sileshi G., Ajayi O., Mweta D. (2007) Synergistic effect of inorganic N and P fertilizers and organic inputs from *Gliricidia sepium* on productivity of intercropped maize in Southern Malawi, Plant Soil 294, 203–217.
- Akinnifesi F.K., Rowe E.C., Livesley S.J., Kwesiga F.R., Vanlauwe B., Alegre J.C. (2004) Tree root architecture, Below-ground interactions in tropical agroecosystems, in: Cadish G., van Noordwijk M., Ong C.K. (Eds.), CAB International, Wallingford, UK, pp. 61–81.
- Bojo J. (1996) The cost of land degradation in sub-Saharan Africa, Ecol. Econ. 16, 161–173.
- Buresh R.J., Tian G. (1997) Soil improvement by trees in sub-Saharan Africa, Agroforest. Syst. 38, 51–76.

- Cassman K.G. (2007) Climate change, biofuels, and global food security, Environ. Res. Lett., doi:10.1088/1748-9326/2/1/011002.
- Chikowo R., Mapfumo P., Nyamugafata P., Giller K.E. (2004) Woody legume fallow productivity, biological N2-fixation and residual benefits to two successive maize crops in Zimbabwe, Plant Soil 262, 303–315.
- Chirwa P.W., Black C.R., Ong C.K., Maghembe J. (2003) Tree and crop productivity in gliricidia/maize/pigeon pea cropping system in southern Malawi, Agroforest. Syst. 59, 265–277.
- Chirwa P.W., Black C.R., Ong C.K., Maghembe J. (2006) Nitrogen dynamics in southern Malawi containing *Gliricidia sepium*, pigeon pea and maize, Agroforest. Syst. 67, 93–106.
- Chirwa P.W., Ong C.K., Maghembe J., Black C.R. (2007) Soil water dynamics in intercropping systems containing *Gliricidia sepium*, pigeon pea and maize in southern Malawi, Agroforest. Syst. 69, 29–43.
- Chirwa T.S., Mafongoya P.L., Chintu R. (2003) Mixed plantedfallows using coppicing and non-coppicing tree species for degraded Acrisols in eastern Zambia, Agroforest. Syst. 59, 243–251.
- Conway G., Toenniessen G. (2003) Science for African food security, Science 299, 1187–1188.
- Dixon J., Gulliver A., Gibbon D. (2001) Farming systems and poverty: Livelihoods in a changing world, FAO and World Bank, Rome, Italy, p. 410.
- FAO (2008) Food and Agriculture Organization of the United Nations statistical database. (http://faostat.fao.org/site/567/) (accessed in January 2008).
- Franzel S., Phiri D., Kwesiga F. (2002) Assessing the adoption potential of improved fallows in eastern Zambia, in: Franzel S., Scherr S.J. (Eds.), Trees on Farms: Assessing the Adoption Potential of Agroforestry Practices in Africa, CAB International, Wallingford, Oxon, UK, pp. 37–64.
- Gacheru E., Rao M.R. (2001) Managing *Striga* infestation on maize using organic and inorganic nutrient sources in western Kenya, Int. J. Pest Manage. 47, 233–239.
- Gathumbi S., Ndufa J., Giller K.E., Cadisch G. (2002) Do species mixtures increase above- and belowground resource capture in woody and herbaceous tropical legumes? Agron. J. 94, 518–526.
- Gowing J.W., Palmer M. (2008) Sustainable agricultural development in sub-Saharan Africa: the case for a paradigm shift in land husbandry, Soil Use Manage. 24, 92–99.
- Ikerra S., Maghembe J., Smithson P., Buresh R. (1999) Soil nitrogen dynamics and relationships with maize yields in gliricidia-maize intercrop in Malawi, Plant Soil 211, 155–164.
- Jama B., Buresh R.J., Ndufa J.K., Shepherd K.D. (1998) Vertical distribution of roots and soil nitrate: Tree species and phosphorus effects, Soil Sci. Soc. Am. J. 62, 280–286.
- Kang B.T., Akinnifesi F.K. (2000) Agroforestry as alternative land-use production systems for the tropics, Nat. Res. Forum 24, 137–151.
- Kang B.T., Reynolds L., Atta-Krah K. (1990) Alley farming, Adv. Agron. 43, 316–359.
- Keil A., Zeller M., Franzel S. (2005) Improved tree fallows in smallholder maize production in Zambia: do initial planters adopt the technology? Agroforest. Syst. 64, 225–236.

- Kuntashula E., Mafongoya P.L., Sileshi G., Lungu S. (2004) Potential of biomass transfer technologies in sustaining vegetable production in the wetlands (*dambos*) of eastern Zambia, Exp. Agr. 40, 37–51.
- Kuntashula E., Sileshi G., Mafongoya P.L., Banda J. (2006) Farmer participatory evaluation of the potential for organic vegetable production in the wetlands of Zambia, Outlook Agric. 35, 299-305.
- Kwesiga F., Akinnifesi F.K., Mafongoya P.L., McDermott M.H., Agumya A. (2003) Agroforestry research and development in southern Africa during the 1990s: Review and challenges ahead, Agroforest. Syst. 59, 173–186.
- Kwesiga F., Coe R. (1994) The effect of short rotation Sesbania sesban planted fallows on maize yield, For. Ecol. Manage. 64, 199–208.
- Li L., Tang C., Rengel Z., Zhang F. (2003) Chickpea facilitates phosphorus uptake by intercropped wheat from an organic phosphorus source, Plant Soil 248, 297–303.
- Mafongoya P.L., Dzowela B.H. (1998) Soil fertility replenishment through agroforestry systems in two contrasting agroecological zones of Zimbabwe, Trans. Zimbabwe Sci. Ass. 72, 31–42.
- Mafongoya P.L., Bationo A., Kihara J., Waswa B.S. (2006) Appropriate technologies to replenish soil fertility in southern Africa, Nutr. Cycl. Agroecosys. 76,137–151.
- Mafongoya P.L., Mpepereki S., Dzowela B.H., Mangwayana E., Mukonesse F. (1997) Effect of pruning quality and method of pruning placement on soil microbial composition, Afr. Crop Sci. Proc. 3, 393–398.
- Makumba W.I.H., Maghembe J.A. (1999) Nitrogen dynamics in two agroforestry technologies practiced by smallholder farmers in the southern part of Malawi, in: Ayuk E., Dery B., Kwesiga F. Makaya P. (Eds.). Proceedings of the 13th Southern African Regional Planning and Review Meeting, 5–11 July 1999, Mangochi, Malawi.
- Mercer D.E. (2004) Adoption of Agroforestry innovations in the tropics: A review, Agroforest. Syst. 61, 311–328.
- Meliyo J.L., Marandu A.E.T., Munuo E. (2007) Evaluation of Gliricidia sepium, Casuarina junghuhniana and Faidherbia albida tree species for improvement of crop production and fuelwood supply in Muheza district, Tanzania, in: Bationo A., Wasaw, B., Kihara J., Kimetu J. (Eds.), Advances in Integrated Soil fertility Management in Sub-Saharan Africa: Challenges and Opportunities, Springer, The Netherlands, pp. 731–736.
- Mweta D.E., Akinnifesi F.K., Saka J.D.K., Makumba W., Chokotho N. (2007). Green manure from prunings and mineral fertilizer affect phosphorus adsorption and uptake by maize crop in a gliricidia-maize intercropping, J. Sci. Res. Essays 2, 446–453.
- Nyamadzawo G., Nyamugafata P., Chikowo R., Giller K. (2007) Residual effects of fallows on selected soil hydraulic properties in a kaolinitic soil subjected to conventional tillage (CT) and no tillage (NT), Agroforest. Syst. 72, 161–168.
- Phiri A.D.K., Kanyama-Phiri G.Y., Snapp S. (1999) Maize and sesbania production in relay cropping at three landscape positions in Malawi, Agroforest. Syst. 47, 153–162.
- Phiri D., Franzel S., Mafongoya P., Jere I., Katanga R., Phiri S. (2004) Who is using the new technology? The association of wealth status and gender with the planting of improved tree fallows in Eastern Zambia, Agroforest. Syst. 79, 131–144.

- Phiri E., Verplancke H., Kwesiga F., Mafongoya P. (2003) Water balance and maize yield following improved sesbania fallow in eastern Zambia, Agroforest. Syst. 59, 197–205.
- Place F., Franzel S., de Wolf J., Rommellse R., Kwesiga F., Niang A., Jama B. (2002) Agroforestry for soil replenishment: evidence on adoption processes in Kenya and Zambia, in: Barrett C.B., Place F., Aboud A.A. (Eds.), Natural Resources Management in African Agriculture: understanding and improving current practices, CAB International, Wallingford, UK, pp. 155–168.
- Rao M.R., Nair P.K.R., Ong C.K. (1997) Biophysical interactions in tropical agroforestry systems, Agroforest. Syst. 38, 1–49.
- Saka A.R., Bunderson W.T., Itimu O.A., Phombeya H.S.K., Mbekeani Y. (1994) The effects of *Acacia albida* on soils and maize grain yields under smallholder farm conditions in Malawi, For. Ecol. Manage. 64, 217–230.
- Sanchez P.A. (2002) Soil Fertility and Hunger in Africa, Science 295, 2019–2020.
- Sanchez P.A., Shepherd K., Soule M.J., Place F.M., Buresh R., Izac A.M. (1997) Soil fertility replenishment in Africa: An investment in natural resource capital, Replenishing soil fertility in Africa, in: Buresh R.J., Sanchez P.A., Cahoun F. (Eds.), SSA Special Publication No. 51, SSA, Madison, Wisconsin, pp. 1–46.
- Sanginga N., Mulongoy K., Swift M.J. (1992) Contribution of soil organisms to the sustainability and productivity of cropping systems in the tropics, Agric. Ecosyst. Environ. 41, 135–152.
- Sauer J., Tchale H., Wobst P. (2007) Alternative soil fertility management options in Malawi: An economic analysis, J. Sustain. Agr. 29, 29–53.
- Schüller M., de Wolf J., Matakala P. (2005) Changing Lives and Landscapes: Beyond the Figures - Results of an Impact Assessment of Agroforestry Technologies in the ICRAF Zambezi Basin Agroforestry Project, World Agroforestry Centre, Maputo, Mozambique.
- Schroth G. (1999) A review of belowground interactions in agroforestry, focusing on mechanisms and management options, Agroforest. Syst. 44, 4–34.
- Sileshi G., Mafongoya P.L. (2003) Effect of rotational fallows on abundance of soil insects and weeds in maize crops in eastern Zambia, Appl. Soil Ecol. 23, 211–222.
- Sileshi G., Mafongoya P.L. (2006a) Long-term effect of legumeimproved fallows on soil invertebrates and maize yield in eastern Zambia, Agric. Ecosyst. Environ. 115, 69–78.
- Sileshi G., Mafongoya P.L. (2006b) Variation in macrofauna communities under contrasting land use systems in eastern Zambia, Appl. Soil Ecol. 33, 49–60.
- Sileshi G., Mafongoya P.L. (2007) Quantity and quality of organic inputs from coppicing leguminous trees influence abundance of soil macrofauna in maize crops in eastern Zambia, Biol. Fertil. Soils 43, 333–340.
- Sileshi G., Akinnifesi F.K., Ajayi O.C., Chakeredza S., Kaonga M., Matakala P. (2007) Contribution of agroforestry to ecosystem services in the miombo eco-region of eastern and southern African, Afr. J. Environ. Sci. Tech. 1, 68–80.
- Sileshi G., Akinnifesi F.K., Ajayi O.C., Place F. (2008) Metaanalysis of maize yield response to planted fallow and green manure legumes in sub-Saharan Africa, Plant Soil 307, 1–19.

- Sileshi G., Kuntashula E., Mafongoya P.L. (2006) Effect of improved fallows on weed infestation in maize in eastern Zambia, Zambia J. Agric. Sci. 8, 6–12.
- Sileshi, G., Mafongoya, P.L., Kwesiga F., Nkunika, P. (2005) Termite damage to maize grown in agroforestry systems, traditional fallows and monoculture on Nitrogen-limited soils in eastern Zambia, Agric. Forest Entomol. 7, 61–69.
- Smale M., Jayne T. (2003) Maize in Eastern and Southern Africa: Seeds of success in retrospect. Environment and

Production Technology Division, Discussion paper No. 97, International Food Policy Research Institute, Washington DC.

- Thangata P.H., Alavalapati J.R.R. (2003) Agroforestry adoption in southern Malawi: the case of mixed intercropping of *Gliricidia sepium* and maize, Agric. Syst. 78, 57–71.
- von Braun J. (2009) Addressing the food crisis: governance, market functioning, and investment in public goods, Food Sec. 1, 9–15.

# **Cereal Landraces for Sustainable Agriculture**

A.C. Newton, T. Akar, J.P. Baresel, P.J. Bebeli, E. Bettencourt, K.V. Bladenopoulos, J.H. Czembor, D.A. Fasoula, A. Katsiotis, K. Koutis, M. Koutsika-Sotiriou, G. Kovacs, H. Larsson, M.A.A. Pinheiro de Carvalho, D. Rubiales, J. Russell, T.M.M. Dos Santos, and M.C. Vaz Patto

## Contents

1	Intro	luction	148
2	Histo	ry of Cereal Landraces	150
3	Diver	sity and Germplasm Collections	151
	3.1	Durum Wheat Landraces Diversity	151
	3.2	Barley Landraces Diversity	152
	3.4	Oat Landraces Diversity	153
	3.5	Maize Landraces Diversity	154
4	Genel	banks and Conservation of Cereal Landraces	154
	4.1	On-Farm Conservation	156
5	Geno	typing and Phenotyping	157
	5.1	Genotyping Technology	158
	5.2	Genotype-Phenotype Association	158
	5.3	Accurate Whole-Plant Field Phenotyping	
		for Exploiting Variation Within Landraces	159
6	Nutri	ent Uptake and Utilisation	160
7	Nutri	tion and Quality	162
8	Biotic	and Abiotic Stress Resistance and Tolerance .	163
	8.1	Wheat Diseases	163
	8.2	Barley Diseases	164
	8.3	Disease Tolerance	166
	8.4	Abiotic Stresses	166
9	Breed	ing: Conversion of Landraces into Modern	
	Cultiv	/ars	168
	9.1	Yield-Based Selection	168
	9.2	Adaptability	168
	9.3	Conversion into Density-Neutral Modern	1.0
	0.4	Cultivars	169
	9.4	Seed Degradation – Cultivar Degeneration	170
10			171
	Partic	supatory Breeding	1/1
11	Partic Legal	Issues	171 173

12	Conclusions	•	 •	 •	 •	•	 •	 •	• •	 •	•	•	•	•	• •	• •	 •	•	174
Re	ferences		 •		 •	•	 •	 •		 •	• •	•	•	•	•	• •	 •		175

Abstract Modern agriculture and conventional breeding and the liberal use of high inputs has resulted in the loss of genetic diversity and the stagnation of yields in cereals in less favourable areas. Increasingly landraces are being replaced by modern cultivars which are less resilient to pests, diseases and abiotic stresses and thereby losing a valuable source of germplasm for meeting the future needs of sustainable agriculture in the context of climate change. Where landraces persist there is concern that their potential is not fully realised. Much effort has gone into collecting, organising, studying and analysing landraces recently and we review the current status and potential for their improved deployment and exploitation, and incorporation of their positive qualities into new cultivars or populations for more sustainable agricultural production. In particular their potential as sources of novel disease and abiotic stress resistance genes or combination of genes if deployed appropriately, of phytonutrients accompanied with optimal micronutrient concentrations which can help alleviate aging-related and chronic diseases, and of nutrient use efficiency traits. We discuss the place of landraces in the origin of modern cereal crops and breeding of elite cereal cultivars, the importance of on-farm and ex situ diversity conservation; how modern genotyping approaches can help both conservation and exploitation; the importance of different phenotyping approaches; and whether legal issues associated with landrace marketing and utilisation need addressing. In this review of the current status and prospects for landraces of cereals in the context

A.C. Newton (🖂)

SCRI, Invergowrie, Dundee DD2 5DA, Scotland, UK e-mail: adrian.newton@scri.ac.uk

E. Lichtfouse et al. (eds.), Sustainable Agriculture Volume 2, DOI 10.1007/978-94-007-0394-0\_10,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Newton et al., Agron. Sustain. Dev. 30 (2010) 237–269. DOI: 10.1051/agro:2009032

of sustainable agriculture, the major points are the following: (1) Landraces have very rich and complex ancestry representing variation in response to many diverse stresses and are vast resources for the development of future crops deriving many sustainable traits from their heritage. (2) There are many germplasm collections of landraces of the major cereals worldwide exhibiting much variation in valuable morphological, agronomic and biochemical traits. The germplasm has been characterised to variable degrees and in many different ways including molecular markers which can assist selection. (3) Much of this germplasm is being maintained both in long-term storage and on farm where it continues to evolve, both of which have their merits and problems. There is much concern about loss of variation, identification, description and accessibility of accessions despite international strategies for addressing these issues. (4) Developments in genotyping technologies are making the variation available in landraces ever more accessible. However, high quality, extensive and detailed, relevant and appropriate phenotyping needs to be associated with the genotyping to enable it to be exploited successfully. We also need to understand the complexity of the genetics of these desirable traits in order to develop new germplasm. (5) Nutrient use efficiency is a very important criterion for sustainability. Landrace material offers a potential source for crop improvement although these traits are highly interactive with their environment, particularly developmental stage, soil conditions and other organisms affecting roots and their environment. (6) Landraces are also a potential source of traits for improved nutrition of cereal crops, particularly antioxidants, phenolics in general, carotenoids and tocol in particular. They also have the potential to improve mineral content, particularly iron and zinc, if these traits can be successfully transferred to improved varieties. (7) Landraces have been shown to be valuable sources of resistance to pathogens and there is more to be gained from such sources. There is also potential, largely unrealised, for disease tolerance and resistance or tolerance of pest and various abiotic stresses too including to toxic environments. (8) Single gene traits are generally easily transferred from landrace germplasm to modern cultivars, but most of the desirable traits characteristic of landraces are complex and difficult to express in different genetic backgrounds. Maintaining these characteristics in heterogeneous landraces is also problematic. Breeding, selection and deployment

methods appropriate to these objectives should be used rather than those used for high input intensive agriculture plant breeding. (9) Participatory plant breeding and variety selection has proven more successful than the approach used in high input breeding programmes for landrace improvement in stress-prone environments where sustainable approaches are a high priority. Despite being more complex to carry out, it not only delivers improved germplasm, but also aids uptake and communication between farmers, researchers and advisors for the benefit of all. (10) Previous seed trade legislation was designed primarily to protect trade and return royalty income to modern plant breeders with expensive programmes to fund. As the desirability of using landraces becomes more apparent to achieve greater sustainability, legislation changes are being made to facilitate this trade too. However, more changes are needed to promote the exploitation of diversity in landraces and encourage their use.

**Keywords** Diversity • Disease • Yield • Quality • Nutrition • Breeding • Genotyping • Competition • Cultivar degeneration•Whole-plant field phenotyping• Non-stop selection•Adaptive variation

## 1 Introduction

Elite cereal cultivars are derived from a relatively narrow germplasm pool and are predominantly well adapted to high input agriculture. However, climate change will bring ever greater challenges in response to both biotic and abiotic stresses. Together with pressures to move towards more sustainable agriculture, there is clearly a need to access and exploit a broader germplasm resource. Cereal landraces are just such a resource which could be very valuable and yet apparently underutilised in contemporary agriculture. Landraces have closer affinity with modern cultivars than wild species and can more easily be used as a foundation material in breeding programmes. During the course of the EU COST Action 860 "Sustainable variety development for low-input and organic agriculture" (2004-2008), a working group on cereal landraces was formed comprising a variety of scientific expertise, linked by common agreement on the potential value of landraces as a resource for contemporary agriculture. In this paper we therefore bring together a review of recent literature on multiple aspects of cereal landraces, offering insights and future direction for their more efficient incorporation and utilisation in agriculture for low-input as well for more favourable environments.

Various definitions of a landrace have evolved since the end of the 19th century. Owing to their complex nature, Zeven (1998) concluded that an all-embracing definition cannot be given. More recently, Camacho Villa et al. (2005) faced with the challenges this creates for inventory and conservation purposes, proposed the following working definition: "a landrace is a dynamic population(s) of a cultivated plant that has historical origin, distinct identity and lacks formal crop improvement, as well as often being genetically diverse, locally adapted and associated with traditional farming systems". They additionally recognised that although the above characteristics are commonly present, they are not always present for any individual landrace.

A distinction is made between landraces and modern, or so called elite, cultivars, the latter being the result of formal crop breeding programmes. It is useful to consider the implications of this from the genetic and breeding points of view. Modern cereal cultivars are almost always bred to be mono-genotypic as inbred or pure lines for self-pollinating species, or one-way hybrids in maize, and thus are genetically homogeneous. They are bred to exploit high-input environments with increased yield levels and with an emphasis on broad or wide adaptation. In the European Union (EU) and other International Union for the Protection of New Varieties of Plants (or UPOV (French: Uion internationale pour la protection des obtentions vtégétales)) countries, modern cultivars are accepted for commercial cultivation after they have passed through the Distinctiveness, Uniformity, and Stability (DUS) and Value of Cultivation and Use (VCU) systems of evaluation trials. In contrast, cereal landraces are genetically heterogeneous. They comprise a large number of distinct homozygous lines in the case of self-pollinating crops or, in the case of cross-pollinators like maize, are populations with more heterozygous components. An important difference between genetic heterogeneity and genetic homogeneity is that the former involves genetic competition among plants, whereas the latter lack such competition (Fasoula and Fasoula, 1997).

Landraces were the principal focus of agricultural production until the end of the nineteenth century with the arrival of formal plant breeding (Harlan, 1975). Then, in more favourable environments, gradual replacement during the early decades of the 20th century by selected component inbred lines and modern cultivars led to their virtual disappearance. However, their cultivation persisted in less favourable environments and despite earlier predictions about their imminent disappearance (Zeven, 1998), landraces still support subsistence farming worldwide. This persistence is not due to increased productivity levels (Almekinders et al., 1994), but because of their increased stability, accomplished through generations of natural and deliberate selection for valuable genes for resistance to biotic and abiotic stresses and inter-genotypic competition and compensation. The components of a genetically diverse population such as a landrace, have a disease buffering effect in reducing the pathogen spread (Frankel et al., 1995). Several important resistance genes were first identified in cereal landraces such as the durable mlo mildew resistance gene in Ethiopian barley landraces (Piffanelli et al., 2004) and later introduced into elite germplasm. In general however, elite cereal germplasm has very few durable resistance genes and little resistance to emerging or increasingly important diseases such as Ramularia collocygni, Fusarium disease complexes and rusts in general, a good example being the new race of stem rust (Puccinia graminis) Ug99. Therefore exploitation of

landrace diversity to identify relevant resistance genes is an important goal. In addition, many cereal landraces are known to possess broad natural variation in valuable neutraceuticals, which has been inadvertently impoverished during the breeding of modern cultivars (Monasterio and Graham, 2000; Murphy et al., 2008).

The important issue of increased yield stability attributed to the inherent heterogeneity of landraces (e.g., Simmonds, 1979) is nicely described by Zeven (1998) as: "Yield stability of landraces under traditional low input agricultural systems is due to the fact that whatever the varying biotic and abiotic stress for each plant, one or more genotypes within the landrace population will yield satisfactorily". This emphasises the urgency for solutions to combat production problems encountered through lack of spatial and temporal uniformity, particularly in stress-prone or marginal environments. A longer-term breeding goal, described below, would include the incorporation of those desirable genes found in different landrace genotypes into one or few superior plants with elevated stability levels (Fasoula, 2008). However, a single genotype must express greater trait plasticity to compensate for its inability to exploit heterogeneity in spatial and temporal dimensions as flexibly as multiple genotypes (Newton et al., 2009).

Awareness that future increases in productivity may depend on improved yields in the high-stress environments, has focussed breeding on specific or narrow adaptation and on the need for conservation of genetic diversity (Cleveland et al., 1999). Conventional plant breeding has been successful in favourable environments and those which can be made favourable through interventions, but it is less successful in lowinput environments, characterized by increased  $G \times E$ interactions, or in organic farming systems. In these cases, participatory breeding approaches with the involvement of local farmers can have a significant and positive influence, as will be discussed. A renewed focus on cereal landraces for breeding purposes could ameliorate some negative consequences of modern agriculture and conventional breeding, such as the liberal use of high inputs, the loss of genetic diversity (Tilman, 1996), and the stagnation of yields in less favourable areas (Annichiarico and Pecetti, 1998).

The role of genebanks and on-farm conservation practices in preserving landrace genetic diversity is increasingly important, particularly since the pivotal involvement of local farmers is continually declining due to the changes brought about by modern agricultural and socio-economic practices. Studies on landrace genetic diversity are a prerequisite for efficient conservation and management, registration purposes and effective use of landraces in breeding programmes. The considerable advances in molecular genotyping and databasing technologies in recent years are beginning to make the variation and resources of landraces more accessible for exploitation. High-throughput genotyping enables genebank accessions with uncertain provenance to be elucidated and thereby enable validation of associated phenotypic data, making them much more useful. In the quest to bridge the phenotype-genotype gap (Miflin, 2000; Parry and Shewry, 2003) and exploit landrace variation, recent advances in genotyping and whole-plant field phenotyping methodology are discussed.

#### 2 History of Cereal Landraces

The Fertile Crescent and Turkey (in Asia Minor) is known as one of the important centres of diversity of many field crops, particularly wheat and barley, and therefore has considerable genetic diversity. Plant domestication from this region over thousands of years has also resulted in the development of enormous diversity. Progressive adaptation to a wide range of environments, responding to various selection pressures including biotic, abiotic and human intervention, has resulted in characteristic intra-specific diversity and differentiation (Teshome et al., 2001) represented by many landraces with specific histories and eco-geographical origins. To better understand the importance of these resources it is necessary to highlight their history.

Archeobotanical data shows that the first domesticated wheat species were einkorn (Triticum monococcum ssp. monococcum) and emmer (Triticum turgidum ssp. dicoccon). These evolved from their wild relatives (T. boeticum and T. dicoccoides respectively) about 10000 years ago (Heun et al., 1997). Both species were the staple food of the human population until the end of the Bronze Age when naked Triticum species became dominant in agricultural lands. In a recent expedition of the Hellenic Genebank, at the edges of the centre of diversity in the northern eastern part of Greece in Thrace natural populations of Triticum monococcum subsp. boeticum and Aegilops speltoides progenitors of the genomes A and B were found in co-existence and in situ conservation has been planned (Kotali, 2008).

Durum wheat (T. turgidum durum) has been of great historical significance, because it provided a range of sub-species that were cultivated widely across the globe for thousands of years (Feuillet et al., 2007). Durum wheat spread out from the Fertile Crescent and through southern Europe, reaching north Africa around 7000 BC (Feldman, 2001). It came into cultivation originally in the Damascus basin in southern Syria about 9800 BC (Zohary and Hopf, 2000). A second route of migration occurred through north Africa during the Middle Ages (Moragues et al., 2006a). The beginning of modern agriculture transformed durum landraces in obsolete 'cultivars' which were gradually replaced by today's elite cultivars where only a fraction of the crop diversity is exploited (Feuillet et al., 2007). Nowadays, wheat landraces are kept

and maintained in germplasm bank collections around the world, but are grown in practice only in the more marginal agricultural environments of their origins.

Bread wheat (T. aestivum L. subsp. aestivum L.) first emerged in cultivated wheat fields approximately 5-6000 years ago (Zohary and Hopf, 2000), resulting from a hybridization between T. turgidum L and T. tauschii L. The D genome of bread wheat originated from Aegilops tauschii and carried alleles adapted to the more continental climate and thus enabled bread wheat to be cultivated over more extensive geographic environments than emmer wheat (Feuillet et al., 2007). The D genome also encodes proteins that restore the softness of the grain endosperm (Chantret et al., 2005) thereby improving bread-making properties. Use for leavened bread production has contributed to its migration to Europe and subsequent widespread cultivation around the world with the development of several different landraces with a diversity of local adaptations driven by different climates and agricultural practices. Hexaploid bread wheat accounts for approximately 90% of world wheat production today (Feuillet et al., 2007).

Barley (Hordeum vulgare L.) also became domesticated in the Fertile Crescent, about 10000 years ago, migrated through Europe, and local agricultural practices and natural selection have led to locally adapted landraces (Jaradat et al., 2004). Valuable information about domestication of cultivated barley, ecogeographical diversity and relevant issues are found in the recent book on barley diversity by Bothmer et al. (2003). In archaeological records, oats, along with rye, were found as a weed contaminant of wheat and barley samples. As a cultivated crop oats was present on a significant scale towards the end of prehistory, mainly in northern European regions (Moore-Colyer, 1995). Domestication of oats is complex, with independent domestication occurring at each ploidy level (Harlan, 1977). Britain was the main area of cultivation of the hexaploid oats. Until the early 1900's cultivated germplasm was landraces adapted to local growing conditions. Nowadays, a large number of modern cultivars trace back to a restricted number of landraces, such as 'Kherson' from which 14 cultivars developed by re-selection and another 80 had 'Kherson' in their pedigrees (Wesenberg et al., 1992).

Maize (Zea mays L. ssp. mays) was domesticated from its wild progenitor teosinte (Zea mays ssp. parviglumis) through a single domestication event, in southern Mexico, between 6000 and 9000 years ago (Matsuoka et al., 2002). After this initial event, introgressions from other teosinte types may have contributed to the maize gene pool and thereby help explain the remarkable phenotypic and genetic diversity (Matsuoka et al., 2002). The same authors suggested also that from early diversification in the Mexican highlands, two paths or lineages of dispersal occurred. One path traces through western and northern Mexico into the south-western USA and then into the eastern USA and Canada. A second path leads out of the highlands to the western and southern lowlands of Mexico into Guatemala, the Caribbean Islands, the lowlands of South America and finally the Andean mountains. There is evidence that these two maize germplasm pools were introduced to Europe at different times and locations. The first introduction came from the Caribbean germplasm, and remained confined to southern Spain probably due to poor adaptation to the mboxEuropean conditions as late maturing populations (Brandolini, 1969). From the 17th century onwards, the north American flint populations, relatively insensitive to day length and with low temperature requirements for flowering, were introduced in northern Europe. Since the introduction of maize to Europe five centuries ago, cultivated populations have evolved under the different selective pressures imposed by the environment and farmers. Adaptation to many environmental niches of European countries for many years explains the large variability and number of landraces which can be observed today (Gauthier et al., 2002).

Clearly landraces have very rich and complex ancestry representing variation in response to many diverse stresses. These are vast resources for the development of future crops deriving many sustainable traits from their heritage. How these resources can be made accessible and exploited will be addressed below.

## **3** Diversity and Germplasm Collections

#### 3.1 Bread Wheat Landraces Diversity

Bread wheat landraces are characterized by their diversity and heterogeneity. However, this genetic diversity needs to be described and measured if it is to be used effectively in breeding and management of plant genetic resources. Traditionally used markers for the description of landrace genetic diversity are morphological and agronomic traits known as descriptors and established by the International Plant Genetic Resources Institute (IPGRI) (IPGRI, 1985). Among them plant height, flowering and anthesis time, spikelet and leaf emergence (Motzo and Guinta, 2007), grain size (Ferrio et al., 2007), grain yield and weight, spikes per unit area (Moragues et al., 2006a,b) and harvest yield index (De Vita et al., 2007) are considered to be the most important ones, and have been successfully used in the phenotyping of bread wheat landraces. Besides IPGRI descriptors other morpho-physiological traits have also been used (Autrique et al., 1996; Nachit et al., 1988; Dencic et al., 2000). Even though morphological descriptors are highly heritable characters and expressed in all environments (Frankel et al., 1995), they are limited in number, and agronomic traits are affected by environmental conditions.

In recent years several physiological, biochemical, molecular and technological traits have been commonly used to characterise bread wheat landraces and to asses their importance as a resource and for food security. Among the biochemical and physiological markers, carbon isotope (13C) discrimination (Ferrio et al., 2007), biomass accumulation (De Vita et al., 2007; Moragues et al., 2006b), storage protein patterns (Gregová et al., 1999, 2004, 2006), High Molecular Weight (HMW) glutenin subunits polymorphism (Caballero et al., 2001) and mineral content (Oury et al., 2006) have been evaluated. Technological parameters, i.e. grain hardness, starch and protein content, viscosity (Igrejas et al., 2002) have been applied also, providing different approaches to the evaluation of landraces diversity and quality. Although biochemical and molecular markers allow fast screening of landrace diversity to detect useful variation for breeding programmes (Gregová et al., 2006) or management of the genetic resources in germplasm collections (Dreisigacker et al., 2005), they are complementary to but cannot replace morphological characterisation (Moragues et al., 2006a; dos Santos et al., 2009; Zeven, 1998). Major difficulties arise if one attempts to interpret molecular or biochemical data generated from germplasm accessions in the absence of their morphological characterisation (Gregová et al., 2006; Zeven, 1998). Overall a significant decrease of genetic diversity has been observed related to the

replacement of bread wheat landraces by elite cultivars which appears to be associated with loss of some quality traits such us protein content and glutenins quality (Gregová et al., 2006; Caballero et al., 2001). At the same there exist enormous gaps in our knowledge that needs to be fulfilled concerning landraces structure, within and among landraces diversity and useful traits.

#### 3.2 Durum Wheat Landraces Diversity

Durum wheat is a primary cereal crop in several regions of the Mediterranean Basin, including the southern peninsular of Italy (Motzo and Guinta, 2007), southern Anatolia of Turkey (Akar and Özgen, 2007) and southern Spain (Ruiz and Martín, 2000). Durum wheat has a great economic importance due to the long tradition of pasta making in Italy (De Vita et al., 2007) and bulgur making in Mediterranean countries (Akar and Özgen, 2007). Archaeological findings suggested that bulgur and cracked wheat could be distinguished as two basic ingredients of Mediterranean cooking and bulgur was known at least since the 3rd millennium BC (Valamoti, 2002). A screen of Madeiran and Canary Islands wheat accessions showed that durum accounts only for 19 and 15% of the germplasm collections respectively (Andrade et al., 2007).

In these and other regions, durum wheat landraces are cultivated by farmers to a very limited extent (Moragues et al., 2006a; dos Santos et al., 2009; Ruiz and Martín, 2000; Zhang et al., 2006; Cherdouh et al., 2005; Teklu et al., 2005; Kebebew et al., 2001; Ben Amer et al., 2001).

As in the case of bread wheat, durum landraces have been replaced in modern farming systems by elite cultivars. The studies performed to evaluate the diversity of durum landraces are predominantly based on germplasm collections that preserve accessions of populations abandoned by farmers. Such studies show the existence of considerable crop heterogeneity and genetic variability (Medini et al., 2005; Masum Akond and Watanabe, 2005; Teklu et al., 2005; Queen et al., 2004; Pagnotta et al., 2004; Alamerew et al., 2004; Kebebew et al., 2001; Ben Amer et al., 2001). Among the DNA molecular markers, Randomly Amplified Polymorhic DNA (RAPD) (Mantzavinou et al., 2005), microsatellites or Simple Sequence Repeats (SSRs) (Zhang et al., 2006; Hao et al., 2006), Amplified Fragment Length Polymorphisms (AFLPs) (Martos et al., 2005) and Restriction Fragment Length Polymorphisms (RFLPs) (Autrique et al., 1996) have been used to genotype durum wheat landraces. However, it is not always easy to determine which material authors have analysed. For example Ruiz and Martín (2000) showed that from 619 durum wheat entries of the Spanish Plant Genetic Resources Center (CRF-INIA (Spanish: Centro de Recursos Fitogenéticos - Instituto Nacional de Investigación Agraria)) collection, 428 were *turgidum* and 126 were durum accessions and claimed that they were representative of durum landrace diversity in southern Spain. However, no detailed information about the origin and sampling methodologies were given.

As a result of the increased economic importance of durum wheat, several breeding programmes aim to develop new cultivars and release old durum landraces (De Vita et al., 2007; D'Amato, 1989). In southern Italy local durum landraces of *Mediterraneum typicum* were cultivated until the beginning of the 1950 when a process of their replacement by modern cultivars began (Motzo and Guinta, 2007; Ruiz and Martín, 2000). As with bread wheat landraces, there has been a significant decrease of genetic diversity in this process which appears to be associated with some loss of quality traits (Oak et al., 2004; De Vita et al., 2007). Overall, it can be concluded that our knowledge on durum landraces need to be improved.

## 3.3 Barley Landraces Diversity

Barley landraces show stability under adverse climatic conditions. This is attributed to the heterogeneity present and that provides them with a buffering capacity. The 'within' as well as 'among' landraces diversity has been the subject of many studies, for example with respect to their agronomic traits, and morphological characters (Assefa and Labuschagne 2004; Abdellaoui et al. 2007). Biochemical markers used for assessing diversity (Kolodinska Brantestam et al., 2003) or complementing morphological data include isozymes (Jaradat and Shahid, 2006) and hordeins (Demissie and Bjørnstad, 1997). Molecular markers used include restriction fragment length polymorphisms (RFLPs) (Bjørnstad et al., 1997; Demissie et al., 1998; Backes et al., 2003), random amplified polymorphic DNA (RAPDs) (Abdellaoui et al., 2007; Manjunatha et al., 2007; Papa et al., 1998) microsatellites or simple sequence repeats (SSRs) (Yahiaoui et al., 2008; Feng et al., 2006; Hamza et al., 2004; Jilal et al., 2008), amplified fragment length polymorphisms (AFLPs) (van Treuren et al., 2006; Assefa et al., 2007) and Inter-Sequence Simple Repeats (ISSRs) (Kolodinska Brantestam et al., 2004).

A controversial issue of genetic diversity studies in landraces is the loss of genetic diversity overtime, as different trends in genetic diversity changes have been observed for different countries. A study carried out on barley germplasm derived from the Nordic and Baltic countries using ISSRs showed that while there were no significant changes of genetic diversity observed during the last century in the northern parts of this geographical region a significant decrease was observed in the southern parts (Kolodinska Brantestam et al., 2004). Similar results were obtained when biochemical markers were employed (Kolodinska Brantestam et al., 2003).

#### 3.4 Oat Landraces Diversity

Oat landraces or oat mixtures were widely grown in Europe until early 1900, while in the USA most cultivars developed up until the 1970s traced back to only seven landraces introduced from Europe (Coffman, 1977). Oat landraces had fairly uniform morphological characters but were heterogeneous for alleles that conditioned reactions to diseases. The 14 cultivars developed as a single-plant selection from the uniform morphologically landrace 'Kherson' referred to above (Coffman, 1977) differed mainly in reactions to diseases and, to a lesser extend, morphological characters. Furthermore, from a single plant selection in a potato field in Cumberland, United Kingdom, the short straw 'Potato Oat' was obtained from which a number of sub-cultivar selections were isolated. We might therefore expect a rather narrow genetic basis to be present but the number of cultivars developed from 'Kherson' and 'Potato Oats' indicates otherwise.

In a number of experiments with many genotypes, morphological characters were shown to possess great phenotypic plasticity in their environmental response (Diederichsen, 2008; Katsiotis et al., 2009). Diederichsen (2008) tested 10105 entries, including landraces, over four years in a single location and found eight environmentally stable characters, while during the RESGEN CT99-106 project (Katsiotis et al., 2009), 1011 entries were tested in four diverse environments over three seasons but only two stable characters were found, namely panicle shape and kernel covering. In both cases the morphological diversity within the oat accessions did not differ between landraces and modern cultivars.

Using molecular markers non-landrace entries showed the same polymorphism as landraces, with the majority of the AFPLP marker variation (89.9%) residing within accessions of each country, revealing the success of oat breeding programmes in maintaining genetic diversity within elite cultivars (Fu et al., 2005). Finally, in a study among Canadian oat cultivars released between 1886 and 2001 using 30 SSRs, a range of increasing and decreasing patterns of allelic changes was observed at different loci and significant allelic decrease was detected in cultivars released after 1970 coming from specific breeding programmes (Fu et al., 2005).

## 3.5 Maize Landraces Diversity

Maize domestication resulted from a single event involving its wild progenitor teosinte (Z. mays subspecies *parviglumis*), introgression from other teosinte types and the segregation into two European germplasm pools (see above) between which much hybridisation occurred. The idea of hybridisation rather than a slow northward dispersion accompanied by selection for earliness is supported in the case of Spain and Portugal where many maize landraces are still cultivated. The Iberian maize germplasm display no close relationship with any American types, sharing alleles with both Caribbean and North American flints (Rebourg et al., 2003; Vaz Patto et al., 2004). Other studies have shown that maize landraces can be distinguished by morphological and agronomic traits (Pinheiro de Carvalho et al., 2008; Brandolini and Brandolini, 2001; Goodman and Paterniani, 1969), biochemical traits such as zeins (de Freitas et al., 2005) or molecular markers (Reif et al., 2005;

Rebourg et al., 2001; Gauthier et al., 2002). More recently SNP markers have been increasingly applied to study useful landraces (Tenaillon et al., 2001). These studies also show that a significant landraces diversity well adapted to agro-ecological conditions still exist in several countries (Pinheiro de Carvalho et al., 2008; Vaz Patto et al., 2007; Brandolini and Brandolini, 2001; Ruiz De Galarreta and Alvarez, 2001).

In summary there are many germplasm collections of landraces of the major cereals worldwide exhibiting much variation in valuable morphological, agronomic and biochemical traits. The germplasm has been characterised to variable degrees and in many different ways including molecular markers which can assist selection.

# 4 Genebanks and Conservation of Cereal Landraces

Throughout the centuries farmers have been the major guardians of genetic diversity. The importance of these crop resources for agriculture and food security was stressed in the International Agricultural Congress at Rome in 1927 (Zeven, 1998). The extinction of traditional farming systems, the aging and exodus of rural population, globalisation, and environmental degradation, have led to extinction of many cereal landraces and much of this diversity has been eroded. As a consequence, during the last century most of this unique cereal biodiversity has disappeared and the information regarding traditional cultivars is presently very scarce. According to FAO (1998), it is estimated that 75% of the genetic diversity of crop plants was lost in the last century. The erosion of these resources results in a severe threat to the world's long-term food security. Although often neglected, the urgent need to conserve and utilize landraces genetic resources as a safeguard against an unpredictable future is evident (Hammer et al., 1999).

The first organised attempts to conserve landrace resources by growing them on farm, (in situ conservation) were made in Austria, during the 1930s (Zeven, 1996). Nowadays the germplasm collections are major guardians of landraces diversity. Increasingly the new tasks of these genebanks are related to the conservation of plant resources and the need to keep accessions representing the landraces diversity and genetic structure (van Treuren et al., 2006). Cereal landraces represents a group of populations, sharing common morphological and agronomic traits, geographical origin and history, and uses (Camacho Villa et al., 2005). These populations are often composed of several genotypes, which together make up the cereal landraces characteristics (Jaradat and Shahid, 2006). For these reasons the genebanks need to collect or conserve rare alleles and avoid genetic drift when accessions are sampled both during field collection and sample regeneration (Mantzavinou et al., 2005). Recently, enormous effort has been made to capture biodiversity being lost by collecting as many germplasm accessions as possible from different geographic regions, especially from the rich centres of diversity. However, it seems that in many cases the sampling strategies were inadequate and the data collected incomplete and scarce, and insufficient attention was paid to ensuring the maintenance of the collected material throughout the lifetime of the genebank reducing the utility of the resource (Sackville Hamilton and Chorlton, 1997). Based on this evidence concern was expressed about the erosion of genetic diversity of the landraces held in the genebanks. It became evident that the management and research on biodiversity requires renewed approaches (Hammer and Gladis, 1996; Hammer and Spahillar, 1998). This problem is increased by the scarcity of knowledge about landrace structure or by the varying understanding of landraces definitions. During the last few years genebank management procedures, such as collecting strategies and techniques, conservation methods, monitoring viability, and regeneration strategies aiming at maintenance of integrity and characterisation of the accessions have been improved by IPGRI (2003). However a major effort is still needed to standardize methodologies used by different genebanks (Engel and Visser, 2003).

A second major problem is the undetermined level of duplicates within and between collections. In order to improve conservation efficiency there is an urgent need to rationalise collections by identifying and minimising unnecessary duplication (Dobrovolskaya et al., 2005) and to develop germplasm core collections, a concept proposed by Frankel and Brown (1984). The first step in the identification of probable duplicates is based on the available passport data (Hintum and Knüpffer, 1995), followed by characterisation data such as agro-morphological, molecular and protein traits. There is an absence of research dealing with this problem and little information is accessible on global information systems. However, there are good examples of research where germplasm collections have been evaluated identifying specific agronomic or quality traits for breeding purposes (Pecetti et al., 2001; Raciti et al., 2003) or assessing accession duplications. For example, Ruiz and Aguiriano (2004) confirmed the existence of 90% of the duplications among 106 cases in durum wheat collections with 266 accessions, using their gliadin patterns. Agro-morphological traits and biochemical or molecular markers are important tools in accession descriptions which will help to fill in information gaps which would otherwise diminish the ability to exploit such material by farmers and plant breeders. In principle, the better managed and more comprehensive a collection, the more valuable it is. The European Cooperative Programme for Plant Genetic Resources (ECP-GR, http://www.ecpgr.cgiar. org/) is a good example of a successful collaboration which should help ensure the continued protection and conservation of cereal landraces through standardised procedures, compatible data documentation systems, and compatibility with European frameworks for better management, study and exchange of resources. Valuable collections of cereal landraces are held at several European and world genebanks (see World Information and Early Warning System (WIEWS) on Plant Genetic Resources for Food and Agriculture (PGRFA), http://apps3.fao.org/wiews/wiews.jsp).

The Convention on Biological Diversity (CBD, 1993) recognizes the contribution of farmers to the conservation and development of genetic diversity. Sharing of benefits and the concomitant increased recognition of the value of the resources are the most effective ways to promote conservation and to ensure the continued availability of plant genetic resources. There is a need to make an economic evaluation of genebank conservation including the issue of benefit sharing but this has not received much attention among those formulating legal measures for the implementation of the CBD. Conservation involves both preservation and evolution. Therefore, ex situ preservation alone cannot provide the lasting benefits that accrue from the conservation of habitats and ecosystems rich in biodiversity (Swaminathan, 2002). This interaction among genebanks and those that are motivated to preserve the traditional seeds will be fundamental to overcoming the problems facing genebanks with seed regeneration, such as genetic drift, contamination and loss.

## 4.1 On-Farm Conservation

Two prevalent methods to conserve plant diversity are in situ and ex situ conservation. In ex situ conservation the genetic resources are conserved outside of their natural habitat (or cultivation territory) in identified genebanks. Unfortunately, in the ex situ conservation methods the variability that has been collected remains static because the natural evolution process is not allowed to continue (Dhillon et al., 2004). In contrast, in situ approaches to conservation are at the level of ecosystems and natural habitats, and include the maintenance and recovery of viable population of species in their natural surroundings, or in the case of domesticated species, in the surroundings where they have developed their distinctive properties. This approach involves two methods, (1) the genetic reserve and (2) on-farm conservation (Hawkes et al., 2002). The former is defined as location, management and monitoring of genetic diversity of natural wild populations within defined areas for active long-term conservation, for example the natural conservation sites in the near-east where the wild Triticum species are conserved in their place of origin. On-farm conservation is the sustainable management of genetic diversity of locally developed traditional crop cultivars along with associated wild and weedy species or forms within traditional agricultural systems. Such in situ methods, including on-farm conservation, have an advantage over ex situ methods since they provide a natural laboratory for evolution to continue and help the continued gradual build-up of traits imparting adaptation to specific ecogeographical regions and those matching the requirements of local tribes, communities and populations. New and more adapted types evolve and thus diversity is augmented. The need for on-farm conservation of original landraces is one of the most important recent questions in plant genetic resource management (Dhillon et al., 2004).

The 'European Plant Conservation Strategy' (Council of Europe, Planta Europea, 2001), the 'European Community Biodiversity Strategy' (European Commission, 2000), the 'Convention on Biological Diversity' (Convention on Biological Diversity, 1992) and the 'International Treaty on Plant Genetic Resources for Food and Agriculture' (http://www. planttreaty.org) all stress the need to improve the efficiency of conservation techniques, particularly those related to in situ conservation of endangered crops and crop wild relatives. However, little progress has been made on the methodologies for on farm conservation of plant genetic diversity (Maxted, 2003), especially in case of cereal landraces. Although there is a general agreement between conservationists on the fact that landraces should to be conserved on farm, i.e. in the place of their natural origin (Kovács, 2006b), in practice this is often very difficult.

In situ conservation programmes were initiated in most European countries (Maxted, 2003), using participatory approaches (see below), but with limited success. The main problems were that the original places are no longer agricultural land, that traditional farming systems and knowledge has disappeared in the given region, or that the in situ conservation is not economic for the farmers and therefore they do not participate in such projects (Kovács, 2006b).

Some very successful in situ conservation strategies have been established in some neighbouring countries. In case of cereals, Turkey, Israel and other countries of the Fertile Crescent have active genetic conservation programmes (Jaradat et al., 2004). One of the first projects was established in Ammiad to conserve wild wheat and wheat and barley diversity in its place of origin (Anikster and Noy-Meir, 1991). However, in Europe there are few such good examples. For example, the AEGRO (AGRI GEN RES 057) project which attempts to promote an integrative approach for the conservation of crop wild relatives and landraces in situ and on farm, such us in the case of Avena landraces. The Scottish landrace protection scheme was set up in 2006 to compile an Inventory of identified landraces and traditional cultivars which are still being grown and used in agriculture in Scotland providing a safety net for the continued use of landraces by storing seed produced by each grower each year (N. Green and G. Saddler, SASA, Edinburgh, personal communication). The farmers are an aging population and the future of Scottish landraces depends on the continued regeneration of the landraces, so if seed harvest fails, the landrace would otherwise be lost.

Clearly much landrace germplasm is being maintained across the world both in long-term storage in major collections and on farm where it continues to evolve, both of which have their merits and problems. There is much concern about loss of variation, identification, description and accessibility of accessions despite international strategies for addressing these issues.

#### 5 Genotyping and Phenotyping

Although the diversity within landraces has been demonstrated to be a powerful means to improve barley yields in marginal environments in recent times (Ceccarelli, 1996), to fully realise and utilise the potential of such resources they need to be accurately and appropriately genotyped and phenotyped and the data made readily available in forms which can be easily interpreted by the plant breeding community.

The genetic structure of landrace collections, when linked to geographical and environmental data, may reveal genomic signatures of selection which are valuable information for breeding for specific environments, farming methods and users. For example, a collection of landraces was made in 1981 in Syria and Jordan from farmers who had been using their own seed for generations (Syrian Jordanian landrace collection (SJLC), Weltzien, 1988). These landraces, sampled from a gradient of agroecological conditions, showed a wide range of responses to drought stress as established by extensive field trials. Preliminary morphological evaluation revealed considerable variation between and within collection sites for many agronomically important characters (Ceccarelli, 1987) and disease reactions (van Leur, 1989). Russell et al. (2003) assayed genetic variability at 21 nuclear and 10 chloroplast microsatellite loci for a stratified subset of 125 barley landrace accessions from the SJLC collection, sampled from five different ecogeographical regions. Chloroplast polymorphism was detected, with most variation being attributable to specific differences between the five regions. A total of 244 nuclear alleles were detected, only 38 of which were common to the five regions sampled, most variation being within sites. There were strong associations between the chloroplast and nuclear SSRs and linkage disequilibrium, the non-random association of alleles, at both linked and unlinked SSR loci, clearly showing that these landraces have 'adapted gene complexes' which might be advantageous for breeding programmes and the genetic diversity and population structure was clearly driven by a drought gradient.

In another study of barley, Bjørnstad et al. (1997), compared cultivated accessions from Europe, north America and Japan with Ethiopian landraces, and found that the Ethiopian germplasm was significantly less diverse than the cultivated germplasm, but that it was also genetically more distinct. Similar studies to those in barley have been carried out in wheat and other cereals. Al Khanjari et al. (2007) surveyed Omani wheats using SSR markers and Stodart et al. (2007) used Diversity Array Technology (DArT<sup>®</sup>) markers to examine 705 accessions from the Australian Winter Cereals Collection and found much diversity, the latter study identifying Nepal as a unique gene pool of particular value.

An exciting development in the potential exploitation of landrace germplasm is the considerable interest in using association-based approaches to identify candidate genes or regions underpinning complex traits (Gaut and Long, 2003; Flint-Garcia et al., 2003; Gupta et al., 2005). Large-scale investigations of sequence variation within genes and across genomes have only just begun for plant species. Such studies are required to determine the distribution and extent of linkage disequilibrium, since this will determine the resolution power of association-based mapping strategies. From studies in other plant species it is clear that the natural decay of linkage disequilibrium with distance occurs at a considerably slower rate in inbreeding systems because effective recombination is severely reduced and genetic polymorphisms remain correlated over longer physical distances (Nordborg et al., 2002; Morrell et al., 2005).

Recently sequence diversity and patterns of linkage disequilibrium were investigated across a 212 kb region in cultivated, landrace and wild barley to determine the impact of inbreeding and evolution history (domestication and selection) (Caldwell et al., 2006). High levels of association were found to stretch across the whole region in the cultivated sample, with linkage disequilibrium values extended across the entire 212kb region. In contrast, linkage disequilibrium and its significance decreased as a function of increasing distance in both landraces and wild barley (Fig. 1). These



**Fig. 1** Plots of linkage disequilibrium as measured with the commonly used statistic  $(r^2)$  (y axis) for each group of pairwise comparisons against the corresponding median distance for cultivated, landrace and wild barley accessions (Caldwell et al., 2006)

contrasting patterns exist despite similar levels of inbreeding and most likely reflect different population histories associated with the occurrence of bottlenecks and selection within the domesticated germplasm. Therefore, large linkage disequilibrium regions in cultivated, low-resolution whole-genome scans could be deployed to identify candidate gene regions; this would then be complemented by fine-scale, high-resolution linkage disequilibrium mapping utilising landraces and wild barley to identify candidate genes.

## 5.1 Genotyping Technology

The developments in association genetics have been facilitated through advances in DNA molecular marker technology reviewed above and by Buckler and Thornsberry (2002). An example of the latest technology is the high throughput Illumina 'Golden Gate Assay' SNP approaches which gives a high density of markers across many genotypes, enabling association genetic approaches to become highly effective. Such surveys provide information on genomic diversity, domestication and evolution, identify geographic regions, which contain high levels of diversity, and discriminate between groups of similar accessions. With the large amount of sequence information in barley, over 400 000 Expressed Sequence tags (ESTs), a platform was set up for high resolution genotyping known as Illumina Oligo Pool Assay (OPA) with 1536 Single Nucleotide Polymorphisms (SNPs) in each assay. Genes which are transcriptionally responsive to abiotic stress were chosen in particular for examining landrace populations.

A subset of the 176 landrace accessions from the SJLC referred to above were subjected to high throughput genotyping on the OPA platform and 72% of the 1536 SNPs were polymorphic and well distributed across the 7 barley chromosomes. Using the available phenotypic data also, whole genome association scans were used to identify and validate genes and markers linked to performance under drought stress (J. Russell et al., personal communication). Again five distinct groups clustering around key ancestors and regions of origin of the germplasm were identified, the accessions from North East Syria and South Jordan being particularly contrasting. Differences in the patterns of diversity between regions of origin were observed along the chromosome, highlighting selection signatures of adaptation to the environment and/or agronomic practices prevalent in these regions.

## 5.2 Genotype-Phenotype Association

Another example of strategies to combat drought stress using landrace germplasm is found in the work of Comadran et al. (2007) who identified barley genomic regions influencing the response of yield and its components to water deficits in a collection of 192



Barley bin map showing significant associations with yield

Fig. 2 Barley chromosome 'bin' map showing the location of phenotypes associated with yield under drought stressed environment conditions

genotypes that represented landraces, old, and contemporary cultivars sampling key regions around the Mediterranean basin and the rest of Europe. They used a stratified set of 50 genomic and EST derived molecular markers, 52 of which were SSRs, and 1131 DArT® markers which together revealed an underlying population sub-structure that corresponded closely to the geographic regions in which the genotypes were grown. The population was phenotyped for yield at two contrasting sites in each of seven Mediterranean sites for two years leading to marker-trait associations to understand the genetic and physiological dynamics underlying barley domestication and intensive breeding carried out in the last century and its relation to adaptation to drought. The yields observed for individual genotypes ranged from 10 t/ha to complete failure of individual genotypes to produce any seed due to the stress. As this was a highly structured sample, after accounting for this in the analysis, multi-environment QTLs were detected most frequently on chromosomes 3H, 4H, 5H and 7H (Fig. 2). One of the encouraging findings of this study was the detection of significant genetic variation for yield in the eight severely

stressed environments where the mean yield was less than 2 t/ha, with one of the most consistent genomic regions being that on chromosome 7H, where four out of the five significant associations came from the Jordanian sites with mean yield ranging from 0.3 to 1.2 t/ha. The detection of QTLs in the low yielding environments offers the prospect of developing Marker Assisted Selection protocols for yield improvement in such situations.

# 5.3 Accurate Whole-Plant Field Phenotyping for Exploiting Variation Within Landraces

In plant breeding literature, the term "phenotypic selection" is often used interchangeably with "visual selection", even though the two are not synonymous. With the current rapid advancements in highthroughput molecular genotyping technologies described above, it is becoming increasingly clear that the limiting factor in applying those powerful technologies to molecular breeding programmes is no more the capacity of genotyping, but the potential for accurate or precision phenotyping (Campos et al., 2004; Fasoula, 2004). Phenotyping becomes a particular challenge when moving from qualitative to quantitative traits, like yield and stability, which have the greatest interest for breeders (Thomas, 2003). A way to bridge the genotype-phenotype gap (Miflin, 2000; Parry and Shewry, 2003) is provided when the unit of evaluation and selection in plant breeding becomes the individual plant (individual genome) and the confounding effects of competition and soil heterogeneity on selection efficiency are addressed with appropriate experimental designs (Fasoulas and Fasoula, 1995, 2000).

Fasoula (2004) demonstrated a methodology for accurate whole-plant field phenotyping using the analysis of crop yield potential, i.e. yield, stability, and responsiveness to inputs (Fasoula and Fasoula, 2002, 2003) which resulted in extracting superior lines from breeder's seed of two local barley (cv. Athenaida) and durum wheat (cv. Kyperounda) cultivars of landrace origin. An excerpt of this analysis is presented in Table 1.

Developments in genotyping technologies are making the variation available in landraces ever more accessible. However, high quality, extensive and detailed, relevant and appropriate phenotyping needs to be associated with the genotyping to enable it to be exploited successfully. We also need to understand the complexity of the genetics of these desirable traits in order to develop new germplasm.

6 Nutrient Uptake and Utilisation

Landraces have developed mostly in environments with low nutrients availability, and may therefore represent a source of variation for selection of varieties adapted to low fertiliser input cropping systems. While the literature on N and P uptake and utilisation of landraces is relatively rich, little has been documented for other nutritional elements. Landraces differ from elite cultivars in their heterogeneous genetic structure as well as for several typical morpho-physiological traits. However, the focus below will be on the morphological and physiological aspects associated with uptake and utilisation of nitrogen and phosphorus by cereal landraces.

The main factor determining nutrient uptake is the root system. This has been shown to be more developed in wheat landraces than in high yielding elite germplasm, especially semi-dwarf varieties (Siddique et al., 1990; Waines and Ehdaie, 2007). Good soil exploration by roots has been shown to be essential for absorption of phosphorus (Gahoonia and Nielsen, 2004a, 2004b) and nitrogen (Cox et al., 1985; Edwards et al., 1990; Feil et al., 1990; Laperche et al., 2006; Wieseler and Horst, 1994), though in the latter case the results are more divergent (Heuberger and Horst, 1995; Kuhlmann et al., 1989; Van Beem, 1997). Thus, appropriately selected landraces with well-developed root systems could be a source of variation for the improvement of nutrient uptake, but its use would require suitable methods for the assessment of the root system, which are still lacking.

 
 Table 1
 Ranking of selected lines within the local Cyprus durum wheat cultivar Kyperounda, of landrace origin, based on the analysis of crop yield potential for accurate whole-plant field phenotyping, using the R-19 honeycomb design, capable to

evaluate 19 entries. The original cultivar Kyperounda (control) was assigned the design code 19 and ranked second last in terms of mean yield per plant

	Yield g/plar	nt	Stabili	ty of performance	Responsivene	ss to inputs	Expected response to selection				
Line code	$\overline{x}$	%	$\overline{x}/s$	%	$(\overline{x}_{sel} - \overline{x})/s$	%	$\overline{x}(\overline{x}_{sel}-\overline{x})/s^2$	%			
7	142.52 <sup>a</sup>	100	2.69	97	1.60	80	4.30	78			
14	$142.12^{ab}$	100	2.28	82	1.76	89	4.01	73			
5	$140.87^{ab}$	99	2.58	93	1.72	87	4.44	81			
17	138.63 <sup>ab</sup>	97	2.78	100	1.78	90	4.95	90			
Other interme	diate lines										
6	123.26 <sup>bcd</sup>	86	2.44	88	1.77	89	4.32	79			
13	$123.09^{bcd}$	86	2.29	82	1.72	87	3.94	72			
3	116.89 <sup>cde</sup>	82	2.15	77	1.70	86	3.66	67			
Kyperounda	$111.00^{cde}$	78	2.03	73	1.67	84	3.39	62			
(19-control)											
8	101.10 <sup>e</sup>	71	2.44	88	1.78	89	4.34	79			

Arbuscular mycorrhizas may considerably increase the active absorbing surface with minor cost for the plant, compared to the formation of roots and root hairs, enhancing P uptake and to a certain extent, uptake of other nutritive elements (Bolan, 1991). The degree of colonialisation has been shown to depend on the host genotype in wheat (Hetrick et al., 1993; Kapulnik and Kushnir, 1991; Manske, 1989, 1990; Manske et al., 1995) and barley (Baon et al., 1993). However, high colonisation rates are not always associated with correspondingly high symbiosis benefits for the plant as this also depends on the host genotype (Hetrick et al., 1993; Manske, 1989, 1990; Manske et al., 1995). There is evidence that certain landraces benefit more from symbiosis than high yielding elite cultivars (Kapulnik and Kushnir, 1991; Manske, 1990). Thus, improvement of the efficiency of symbiosis based on selected landraces might be possible.

Nitrogen-fixing bacteria in the rhizosphere are also important root symbiotic relationships for nutrient uptake, especially in Azospirillum species. These associations are also influenced by the host genotype and are particularly developed in several wheat landraces and wild ancestors originating from the southern Mediterranean basin (Kapulnik et al., 1983, 1985, 1987). Although the bacteria contribute to N nutrition of the host plants, the major benefit for the plant was from stimulation of root growth and thus drought tolerance (Kapulnik et al., 1983, 1985, 1987).

Landraces and old varieties are often later maturing than modern cultivars, especially those bred for dryer environments (Canevara et al., 1994). This may be of importance for the uptake of N in N-limited environments. If little or no N fertiliser is applied, N supply depends on mineralisation of soil organic matter, organic fertilisers and crop residuals. The time course of mineralisation is not always compatible with that of the crop's requirements (Panga and Lethaya, 2000) and varies greatly from year-to-year due to factors such as the weather and preceding crops. The potential uptake by cereal plants is mostly higher than the actual mineralisation in the soil, especially in the later growth stages (Baresel et al., 2008). Late-maturing genotypes may consequently absorb more N overall, and may therefore be better adapted to these conditions, if water availability is not limiting.

Under N-limited conditions, wheat landraces and varieties with a taller growth habit and lower harvest index have been shown to absorb and translocate more



**Fig. 3** Protein yields of modern varieties and breeding lines, varieties bred for organic farming and landraces or very old varieties in an environment with high and low average protein yield, respectively. The latter group have relatively high protein yields in the low-yielding environments, but low performance in the high-yielding environments. Results of a linear regression analysis on 70 genotypes in 9 environments; see Baresel et al. (2005) for details

nitrogen into the grain than modern cultivars (Baresel et al., 2005). Figure 3 shows that old landraces and very old cultivars with similar morphological habit may absorb much more nitrogen under low-yielding conditions than modern cultivars and breeding lines. One reason might be greater pre-anthesis uptake and buffering capacity in genotypes with high vegetative biomass (Baresel et al., 2008), but this aspect has been little investigated. Nitrogen absorbed before grain filling is remobilised and then translocated to the grain after anthesis. Translocation efficiency might therefore also contribute considerably to efficient N utilisation. However, only small differences between genotypes could be detected in the absence of leaf diseases (Bertholdsson and Stoy, 1995; Johnson et al., 1967; Papakosta and Garianas, 1991; Pommer, 1990) and therefore genetic variation of translocation efficiency does not appear to offer much opportunity to improve N efficiency.

There are many possible combinations of environmental factors which determine nutrient availability and landraces generally reach their genetic equilibrium in environments with reduced nutrient availability and variable conditions from year to year. They therefore show many different adaptive types (Attene and Veronesi, 1991) and if maintained through on-farm conservation, will offer a valuable resource for finding ways of improving nutrient use efficiency without loosing genetic plasticity.

Nutrient use efficiency is a very important criterion for sustainability. Landrace material offers a potential source for crop improvement although these traits are highly interactive with their environment, particularly developmental stage, soil conditions and other organisms affecting roots and their environment.

#### 7 Nutrition and Quality

Early landraces and wild species provide a broad representation of natural variation not only in agronomically important traits, but also in nutraceuticals, which have decreased during the breeding of modern cultivars. The nutritional status of the most important staple foods, such as the cereals is ultimately dependent on their metabolic composition (Galili et al., 2002). Nevertheless, while traits associated with yield and resistance have been the focus of most research, quality traits that are dependent on chemical composition are less well studied. There are some notable exceptions, such as protein content and structure of wheat cultivars and landraces (Láng, 2006; Rakszegi et al., 2006). Mineral content in modern wheat cultivars has significantly decreased including copper, iron, magnesiou, manganese, phosphorous, selenium, and zinc (Murphy and Jones, 2006). Looking for breeding sources for higher levels of iron and zinc has revealed that the highest levels can be found in landraces and low yielding genotypes (Monasterio and Graham, 2000). An ancestral wild wheat gene was found that accelerates senescence and increases nutrient remobilisation from leaves to developing grains. The gene was found in all wild emmer accessions and most domesticated emmer but not in durum lines or hexaploid wheat (Uauy et al., 2006). Cereal seed mineral content have been analysed in a wide range of cereal genebank accessions (Bálint et al., 2003) to find optimal sources to increase mineral content of modern wheat cultivars. Emmer wheat is a promising source of genetic variation for protein, zinc and iron content.

In bread wheat, the concentration of carotenoids is low, but they are more abundant in, for example, durum wheat, emmer and einkorn cultivars and landraces, having higher concentration in landraces then in cultivars (Panfili et al., 2004). Tocols, in contrast, are abundant both in bread wheat and durum wheat (Panfili et al., 2004). As carotenoid and tocol contents are independent traits (Hidalgo et al., 2006) that were not subject to conscious human selection and are selectively neutral, the wide range of natural variation still exists in the cereal genetic resources and landraces. In recent experiments bread wheat landraces, emmer and einkorn were found to be the best sources of tocols, while durum landraces and emmer were the best sources of carotenoids (Hidalgo et al., 2006).

Total phenolics in wheat varied both with cultivar and farming site (Gélinas and McKinnon, 2006). The phenolic compounds flavonoids, saponins, lignans and sterols are found in oat grain, but in minor quantities. Their concentrations are very low compared to avenantramides and tocols but have antioxidant and other bioactivitie properties (Peterson, 2004). Tocopherols and tocotrienols have higher concentrations in cultivars with high total lipids (Bryngelsson et al., 2002) and landraces of black oats have higher concentrations of antioxidants than elite cultivars (Mannerstedt-Fogelfors, 2001).

Maize contains appreciable amounts of carotenoids (Wurtzel, 2004) and has a greater total phenolic content and total antioxidant activity than wheat, oats or rice (Adom and Liu, 2002). In the maize kernel, tocopherol and oil content may be physiologically associated (Kurilich and Juvik 1999) while the natural variation detected in maize landraces kernel pigmentation may be associated with increased availability of certain antioxidant compounds. White maize polyphenolics have shown to have antioxidant and anti-carcinogenic effects (Del Pozo-Insfran et al., 2006). Blue, purple and red-pigmented maize kernels are also rich in anthocyanins with well-established antioxidant and bioactive properties (Del Pozo-Insfran et al., 2006). Purple maize, that has been cultivated for centuries in the Andean Region, is a good example of this (Pedrechi and Cisneros-Zevallos, 2006).

In general, cereal landraces and old varieties are among the best sources of phytonutrients accompanied with optimal micronutrient concentrations. Grains, fruits and vegetables contain a broad variety of phytonutrients, which show a significant effect on reducing the incidence of aging-related and chronic diseases. Among the numerous antioxidant compounds present in these foods, grain fat-soluble antioxidants and their unique bioactive compounds play an important role in disease prevention. The additive and synergistic effects of such phytochemicals in fruits, vegetables and whole grains are thought to be primarily responsible for their health benefits.

Landraces are clearly a potential source of traits for improved nutrition of cereal crops, particularly antioxidants, phenolics in general, carotenoids and tocol in particular. They also have the potential to improve mineral content, particularly iron and zinc, if these traits can be successfully transferred to improved varieties.

# 8 Biotic and Abiotic Stress Resistance and Tolerance

The replacement of landraces by homogeneous cultivars entails a significant loss of genetic variation for resistance to biotic and abiotic stresses. Elite cultivars may not possess the combined resistances already present in the landrace that they are intended to replace. Also, landraces might be a good reservoir of resistance mechanisms other than the hypersensitive mechanism typically exploited in modern cultivars due to its simple inheritance and complete expression. Most of the studies cited and others are either screening for qualitative resistance or more detailed mapping of the major genes responsible. There are few studies of landraces for partial and polygenic resistance due to the increased resources needed to obtain quantitative data. However, such studies are needed to determine whether multiple sources of partial resistance from such sources offer genes more likely to be durable than those in the elite gene pools. Below we will review the biotic stresses for just wheat and barley where landraces have provided valuable sources of resistance, then disease tolerance and some abiotic stresses for cereals in general.

#### 8.1 Wheat Diseases

#### 8.1.1 Septoria Leaf Blotch

The fungus *Mycosphaerella graminicola* (Fuckel) Schr. (sexual stage of *Septoria tritici* Rob. in Desm) or Septoria Tritici Blotch (STB) is currently a major disease of worldwide distribution. Most currently grown wheat cultivars are more or less susceptible to *M. graminicola*. The Italian landrace Rieti, an ancestor of many modern European wheat cultivars has been identified as very resistant to all studied isolates (Arraiano and Brown, 2006). The presence of *Stb6* gene in both European and Chinese landraces suggests that this gene has been present in cultivated wheat since the earliest times of agriculture (Chartrain et al., 2005). Resistance has also been found in Czech and Slovak landraces (Vechet and Vojácková, 2005).

#### 8.1.2 Powdery Mildew

Powdery mildew of wheat is a foliar disease caused by the obligate biotrophic fungus *Blumeria graminis* (DC.) Speer f. sp. *tritici* (syn. *Erysiphe graminis* DC. f. sp. *tritici* G. Marchal) that can cause loss in both grain yield and quality. Screening of old wheat cultivars, landraces and related species for resistance to powdery mildew started in the 1930's (Hsam and Zeller, 2002) and *Pm* genes have since been identified in many different, widely distributed wheat cultivars and landraces. Non-major gene resistance has been sought such as the durable adult plant resistance found in the landrace accession k-15560 (Peusha et al., 2002).

#### 8.1.3 Fusarium Head Blight

There is little resistance available to the major disease Fusarium Head Blight caused by *Fusarium graminearum* Sch. although some has recently been identified in landraces (Zhang et al., 2000).

#### 8.1.4 Bunts and Smuts

Common bunts (*Tilletia foetida* (Wall.) Liro and *T. caries* (DC.) Tul.) are important diseases that are easily controlled by seed dressing, but might become more important in organic and low input agriculture. More than 15 resistance genes (Bt1-15) have been identified in wheat and the landrace PI178383, originally collected in Turkey, carries resistance genes Bt-8, Bt-9 and Bt-10 plus an unidentified factor (Goates, 1996). Resistance has also been reported in landraces of bread wheat (Hubert and Buertsmayr, 2006)

and of durum wheat (Mamluk and Nachit, 1994). Karrnal bunt (*Tilletia indica* Mitra) is the most recently described smut of wheat and resistance has been identified in Indian landraces (Anon, 1943).

#### 8.1.5 Rust Diseases

Stem rust (Puccinia graminis Pers.) resistance transferred to bread wheat from Yaroslav emmer (Sr2 complex) in combination with other genes seems to have provided the foundation for durable resistance to stem rust in CIMMYT germplasm (Roelfs, 1988) in the last 50 years. Sources of durable resistance to stem rust in durum wheat have been reported, like the durum wheat Glossy Huguenot, effective in Australia over the past 100 years (Hare, 1997). Interestingly, its resistance is also quantitative and based on a reduced number of pustules in adult plants and a delayed onset of disease. As indicated by Roelfs (1988), of the 41 known genes for stem rust resistance, 20 originated in species other than T. aestivum and T. turgidum; of the 35 known genes for leaf rust resistance, 12 originated in species other than T. aestivum and T. turgidum. Among the genes originating from T. aestivum for resistance to either rust, a number of these are from landraces (McIntosh et al., 1998). The recently reported spread of the stem rust race Ug99 and the dependence of so many elite cultivars of wheat on the Sr31 resistance gene has spurred renewed interest in surveying landrace collections for novel resistance sources (Bonman et al., 2007).

Durable resistance to leaf rust (Puccinia recondita f.sp. tritici (Eriks. & E. Henn.) D.M. Henderson and P. triticina Erikss.) of wheat is thought to be more difficult to obtain than to stem rust but resistance against leaf rust has been identified that appears more durable than the norm. Resistance in the bread wheat cultivars Americano 44D, and Frontana, derived from resistant landraces, appears to be particularly durable. It appears that Lr12 and Lr13, both genes for hypersensitive resistance that is expressed only in the adult plant stage, in combination with Lr34, are the basis of most of this resistance (Roelfs, 1988; Rubiales and Niks, 2000). Americano 44D, a Uruguayan landrace of unknown origin (called Universal 2 in Argentina), was used by Klein in breeding early Argentinean lines and is now considered another important source of durable resistance to leaf rust (Van Ginkel and Rajaram, 1992).

Durable resistance to yellow rust (*Puccinia striiformis* Westend.) has been described in wheat landraces from China, Italy and the Netherlands (Van Dijk et al., 1988; Zhang, 1995). Partial resistance combined with temperature-sensitive resistance have been suggested as the major components of the durable resistance found in old winter wheats in the Netherlands (Van Dijk et al., 1988). Sources of partial resistance to leaf rust have been recently reported in landraces from various origins (Fekadu and Parlevliet, 1997; Martínez et al., 2001a, b; Shtaya et al., 2006a, b).

Histological studies on interactions between plants and rusts can help both to discern the various resistance mechanisms and to combine them in a genotype in the hope to increase durability. The commonly used hypersensitivity resistance, typically conferred by single genes with race-specific effectiveness, is due to a post-haustorial defence mechanism. This type of resistance is very common in non-host interactions (Niks and Rubiales, 2002). It also is the mechanism responsible for the partial resistance of some wheat landraces to wheat leaf rust (Martínez et al., 2001c).

#### 8.1.6 Aphids

Resistance to the Russian wheat aphid (*Diuraphis noxia* Kurd.) has been found in landraces of wheat from Iran and the former Soviet Union (Du Toit, 1987).

#### 8.2 Barley Diseases

#### 8.2.1 Powdery Mildew

In most barley growing regions powdery mildew (*Blumeria graminis*, (syn.: *Erysiphe graminis*) (D. C. Speer) f. sp. *hordei*) is very common. Intensive studies were carried out on barley landraces from Ethiopia (Negassa, 1985a, b), Jordan and Syria (van Leur, 1989) and other countries of the Near East (Weltzien, 1988), Europe (Honecker, 1938), India (Freisleben, 1940), Japan (Hiura, 1960) and world-wide (Moseman, 1955; Nover and Mansfeld, 1955, 1956; Hoffmann and Nover, 1959; Rigina, 1966; Wiberg, 1974a, b; Moseman and Smith, 1976;
Czembor, 2002). The main history of incorporation of powdery mildew resistance genes in cultivated barley and the exploration of their genetic diversity in Europe is described by Wolfe and Schwarzbach (1978).

The situation regarding diversity of genes for partial resistance to powdery mildew in cereal landraces is less clear. However, many assessments of partial resistance to pathogens in barley have been made with powdery mildew (Wright and Heale, 1984; Asher and Thomas, 1983, 1984, 1987; Anderson and Torp, 1986; Carver, 1986; Heun, 1986; Geiger and Heun, 1989; Newton, 1990; Kmecl et al., 1995).

#### 8.2.2 Rust Diseases

Three rust species commonly occur on barley: leaf rust (*Puccinia hordei* Otth), stripe rust (*P. striiformis* West. f. sp. *hordei*) and stem rust (*P. graminis* Pers.: Pers. f. sp. *tritici* Eriks. et Henn.).

The origins of resistance sources were similar to those of mildew-resistant landraces from the Mediterranean region where both the host and the pathogen are indigenous and have co-evolved (Anikster and Wahl, 1979). Israel in particular is part of the centre of origin and genetic variation of wild native Hordeum species H. vulgare ssp. spontaneum Koch, H. bulbosum L. and H. murinum (Wahl et al., 1988; Kandawa-Schulz, 1996). Some interesting material has also been found in Azerbaijan and Turkmenia (Bakhteev collection) and Iran (Kuckuck collection) (Nover and Lehmann, 1974; Walther and Lehmann, 1980). Many studies described the activities in evaluation and in breeding for major genes and partial resistance (Clifford, 1985; Reinhold and Sharp, 1986; Yahyaoui et al., 1988; Khokhlova et al., 1989; Jin et al., 1995; Lukyanova and Terentyeva, 1997; Alemayehu and Parlevliet, 1997). Resistance against a new pathotype of Puccinia hordei with virulence for the resistance gene Rph7 has been identified in barley landraces (Shtaya et al., 2006c).

Stripe rust is known in most of the barley growing regions. Screening for new sources of resistance has been carried out by many groups (Nover and Lehmann, 1966, 1970, 1975; Upadhyay and Prakash, 1977; Stubbs, 1985; Van Leur et al., 1989; Okunowski, 1990; Luthra et al., 1992; Hill et al., 1995). As with wheat rusts, much durable resistance is pre-haustorial and is known to be the mechanism responsible for the partial resistance of some barley landraces to barley leaf rust (Shtaya et al., 2006a, b). Sources of partial resistance to leaf rust (*P. hordei*) have been recently reported in barley and wheat landraces from Spain and from Fertile Crescent (Martínez et al., 2001a, b; Shtaya et al., 2006a, b).

#### 8.2.3 Scald

Evaluations of barley for resistance to scald (*Rhynchosporium secalis* (Oudem.) J. J. Davis *f. sp. hordei*) have been carried out in many countries (Fukuyama et al., 1998; Yitbarek et al., 1998). Recent examples of such reports are: novel alleles at the *Rrs1* and other loci have been found for *Rhynchosporium secalis* resistance (Grønnerød et al., 2002; Bjørnstad et al., 2004).

#### 8.2.4 Net Blotch

Studies of landraces for resistance to net blotch (*Pyrenophora teres* (Died.) Drechsl. f. *teres*) have been carried out by many scientists (Schaller and Wiebe, 1952; Buchannon and McDonald, 1965; Gaike, 1970; Smirnova and Trofimovskaya, 1985; Proeseler et al., 1989; Lukyanova, 1990; Faiad et al., 1996). Sato and Takeda (1994) studied the variation of host resistance of 2233 accessions of the barley world collection and found sources of resistance in accessions from Ethiopia, North Africa and Korea. New sources with resistance to up to eight races of *P. teres* were found among Peruvian landrace accessions (Afanasenko et al., 2000).

#### 8.2.5 Barley Stripe

Many studies were conducted to identify new sources of resistance to barley stripe (*Pyrenophora graminea* Ito & Kuribayashi) including landraces (Baigulova and Pitonya, 1979; Nettevich and Vlasenko, 1985; Skou and Haahr, 1985; Van Leur et al., 1989; Su et al., 1989; Lukyanova, 1990; Bisht and Mithal, 1991; Ceccarelli et al., 1976; Kirdoglo, 1990; Skou et al., 1992, 1994).

#### 8.2.6 Common Root Rot and Spot Blotch

Several germplasm collections have been evaluated and resistance to common root rot and spot blotch (*Cochliobolus sativus* (Ito & Kurib.) Drechsler ex Dastur) has been identified from several resources (Banttari et al., 1975; Velibekova, 1981; Rochev and Levitin, 1986; Lehmann et al., 1988; Lukyanova, 1990; Gilchrist et al., 1995; Semeane, 1995; Faiad et al., 1996).

### 8.2.7 The Smuts

Three species of cereal smut attack barley: *Ustilago nuda* (Jens.) Rostr. (*U. segetum* var. *nuda*), *U. ni-gra* Tapke (*U. segetum* var. *avenae*) and *U. hordei* (Pers.) Lagerh. (*U. segetum*). Many studies on resistance to these pathogens were conducted (Shchelko, 1969, Nover et al., 1976; Damania and Porceddu, 1981; Onishkova, 1987; Dunaevskij et al., 1989; Surin, 1989; Lukyanova, 1990; Dubey and Mishra, 1992) identifying sources of resistance from landraces from Ethiopia, Yemen, Tibet, Canada and USA.

#### 8.2.8 Fusarium Complexes

Many scientists described differences in fusarium resistance between cultivars (Grigor'ev et al., 1988; van Leur, 1989; Gu, 1989; Corazza et al., 1990; Khatskevitch and Benken, 1990; Lukyanova, 1990; Takeda, 1992; Filippova et al., 1993; Nelson and Burgess, 1994; Perkowski et al., 1995, 1997). Based on these reports it can be concluded that valuable sources of resistance were identified in the East Asian region landrace accessions in particular (Takeda and Heta, 1989).

#### 8.2.9 Viruses

Yasuda and Rikiishi (1997) evaluated a total of 4342 barley accessions from the world collection on a field in Japan infected with strain I (Kashiwazaki et al., 1989) of BaYMV for resistance. The percentage of asymptomatic cultivars was highest among Ethiopian landraces followed by those from Japan. Cultivars showing severe disease symptoms were frequently found among Chinese, Nepalese, southeast Asian, north African, north American and European accessions. Field resistance to barley yellow dwarf was detected in several Ethiopian barleys (Schaller et al., 1964).

#### 8.2.10 Aphids

The spring two-rowed barley RWA 1758 has been developed via selection from CIho 4165, a landrace originally collected in Afghanistan (Bregitzer et al., 2008).

#### 8.3 Disease Tolerance

A character much neglected in elite breeding programmes is disease tolerance, not least because of the varied definitions of the term and the difficulty of measuring it (Bingham and Newton, 2009). However, interspecific variation has been found in cereals, for example powdery mildew-infected leaves of a wild oat showed a smaller reduction in net photosynthetic rate than a cultivated oat genotype under comparable infection severities (Sabri et al., 1997). The wild oat leaves also showed a slower rate of disease-associated senescence. In a comparison of wild and cultivated barley genotypes similar effects were reported (Akhkha et al., 2003) but the effect of this variation on tolerance at the scale of the crop canopy has not been determined. Several landrace accessions appear amongst the accessions screened for tolerance in barley (Newton and Thomas, 1994; Newton et al., 1998, 2000), though not disproportionately with more modern cultivars.

## 8.4 Abiotic Stresses

Generally the genus *Hordeum* shows a high degree of adaptation to different stressful environments.

#### 8.4.1 Drought Tolerance

Drought is the most common abiotic constraint for stable barley production in rain-fed areas. Under Mediterranean conditions, water stress is particularly common at the end of barley life cycle (Passiuora, 1996). In comparison to other cereals, barley is well adapted to arid environments and the immediate progenitor of cultivated barley *H. vulgare* ssp. *spontaneum* can grow in desert condition (Nevo, 1992; Zohary and Hopf, 1998). Such ecotypes were identified in desert locations in Jordan (Jaradat et al., 1996). The study of drought stress on yield in Mediterranean environments noted above (Comadran et al., 2007) identified genomic regions in landraces that may be very valuable for combating such stress.

#### 8.4.2 Frost Tolerance

Selection of highly frost tolerant lines from Turkish barley landraces has resulted in conversion of spring based production into winter based and enlargement of winter sown barley production in the Turkish highlands since 1940. Today 60% of barley production has been provided by winter sown barley, yield has doubled and these landraces have been routinely and widely used as parents in many breeding programmes in Turkey (Akar et al., 2009).

#### 8.4.3 Salinity Tolerance

In the investigation of Abo-Elenin et al. (1981) 1163 entries were tested in the field and 777 in lysimeters. In this study 'Abyssinia' was the most tolerant. Mano et al. (1996) screened 6712 accessions for salt tolerance at germination. Accession variation showed a normal distribution and the most tolerant ones could germinate in sea water. However, six-rowed cultivars were more tolerant than two-rowed, hull-less than hulled, normal than semi-dwarf '*uzu*', and winter than spring. In another study Mano et al. (1995) evaluated 5182 barley cultivars for salt tolerance at seedling stage. Generally the geographical differentiation among tested accessions was not clear.

Tolerance to salinity is more frequent in bread than is durum wheat, as bread wheat has a salt-exclusion mechanism but durum does not. This was found in landraces originating from saline areas of the Middle East and is conferred by *Nax* genes which therefore could be incorporated into both durum and bread wheat (Munns, 2005).

## 8.4.4 Acid and Alkaline Soils and Tolerance to Metal Toxicity

Cereal adaptation to acid and alkaline soils is limited by two major problems, the aluminium and manganese toxicity in acid environments and boron toxicity in alkaline ones. The soil acidity is a serious agricultural problem, affecting as much as 40% of the world's arable land and up to 70% of the world's potentially arable land (Kochian et al., 2005; Hede et al., 2001). Aluminium toxicity is a main growth and yield-limiting factor on soils with pHs below 5.0 (Davies, 1994), and can directly reduce yield by up to 60% (Tang et al., 2003). Amelioration of soil surface layer is not a reasonable solution in low input and organic agriculture, and because plant roots develop in lower acid layers to reach critical water and nutrient supplies. Selection and development of genotypes with enhanced tolerance to acid soils and toxic levels of aluminium is considered to be a more effective solution to this problem.

Cereal crops show very different responses to aluminium toxicity and soil acidity. The highest aluminium tolerance is detected among rye (Little, 1988), followed by oat (Slaski, 1992), wheat (Aniol and Madej, 1996), barley (Foy et al., 1965), and corn (Horst et al., 1997). Some experimental evidence shows a dramatic variation in aluminium tolerance among cultivars, which can be related to their genetic variability (Carver and Ownby, 1995). Landraces are important sources of this variability to improve the aluminium and acid soils tolerance in breeding programmes, but most research on aluminium tolerance has been carried out on elite cultivars or isogenic lines (Kochian et al., 2005) and there are very few studies to identify landrace tolerance for these traits (Pinheiro de Carvalho et al., 2003; Gudu et al., 2001; De Sousa, 1998; Cosic et al., 1994). Forty-eight accessions representing 16 Madeiran wheat landraces were screened for their aluminium tolerance using erichrome staining and root elongation (Pinheiro de Carvalho et al., 2003, 2004) and the accumulation of callose in the root types (dos Santos et al., 2005). The variability of landrace responses to the presence of aluminium and the existence of high performing accessions with better performance has been shown by comparison with elite cultivars such as Maringa. Durum wheat landraces show less variability and are moderately sensitive or tolerant to the presence of aluminium (Pinheiro de Carvalho et al., 2003; Cosic et al., 1994). De Sousa (1998) published a classification of aluminium tolerance of 76 wheat cultivars, including several landraces introduced in earlier twentieth century. The major sources of aluminium tolerance in wheat are considered to be originated from Brazil (Zhou et al., 2007). However, Stodart et al.

(2007) and Zhou et al. (2007), through the screening of wheat accessions from different countries showed the existence of potential new sources of aluminium resistance among the landraces germplasm originated from Bulgaria, Croatia, India, Italy, Nepal, Spain, Tunisia, and Turkey. The sources of aluminium tolerance in barley are limited to old cultivars and landraces, and represent multiples alleles of a single locus (Nawrot et al., 2001). The evaluation of corn germplasm, in two different studies screening of 76 accessions of unknown number of Kenyan maize landraces (Gudu et al., 2001) and 40 accessions of five Madeiran maize landraces (Pinheiro de Carvalho et al., 2004) also showed their high variability in aluminium tolerance, with several accessions presenting better performance then commercial standards.

The soil alkalinity also affects agricultural crops, growing in arable soils with pHs between 8 and 10. In barley, boron toxicity is directly responsible for yield penalties of up to 17% (Cartwright et al., 1984). The screening of 444 accessions of winter barley and 19 accessions of durum wheat, including landraces from Europe, west Asia and north Africa showed that boron tolerance is associated with geographic origin (Yau, 2002; Yau et al., 1995).

Landraces have long been assumed to be valuable sources of resistance to pathogens and the literature demonstrates that there is much to be gained from such sources. Transfer of resistance genes from landraces to modern cultivars is likely to be less problematic than from wild accessions. There is clearly also potential, largely unrealised, for disease tolerance and resistance or tolerance of pest and various abiotic stresses too including to toxic environments.

# 9 Breeding: Conversion of Landraces into Modern Cultivars

In this section the advantages of breeding cereal landraces, the methodology of improvement and modifications of breeding assumptions will be proposed.

# 9.1 Yield-Based Selection

Breeding from landrace accessions is a strategy being used to improve yield and yield stability in less favourable agricultural system with lower input levels. Heritabilities are higher in more favourable than poor environments (Blum, 1988). The stagnation of yields in these areas (Annicchiarico and Pecetti, 1993) is mainly related to the narrow genetic base of the more recently bred, high-yielding cereals (Pecetti et al., 2002). Two different approaches may be followed to raise yields in the long-term: one is based on increasing yield potential of broadly adapted cultivars, while the other relies on the better exploitation of the adaptive features of genotypes by fitting cultivars to specific target environments (Acevedo and Fereres, 1993). Improvement of grain yield potential in small-grain cereals has traditionally relied on direct selection for this trait (Annichiarico and Pecetti, 1998). Traditional breeding is based on a combination of bulk-pedigree method of selection, applied selection in the presence of stress, and use of adapted germplasm (Ceccarelli and Grando, 1997).

The opportunity to complement traditional breeding with use of indices of indirect selection for yield including sets of morpho-physiological traits, also known as an analytical breeding, has been put forward, especially for less favourable regions (Richards, 1982). There are two strategies in analytical breeding for identifying morpho-physiological traits usable as tools for selection (Fischer, 1981; Jackson et al., 1996). The first, called the 'black box' strategy, consists of assessing a germplasm pool for correlated response to yield gain deriving from selection for sets of putatively useful traits. The second, defined as the 'ideotype strategy', is based on the assessment of traits chosen a priori, through comparison in isolines or prediction of performance in crop growth models of different trait levels (Annichiarico and Pecetti, 1998).

## 9.2 Adaptability

In landraces an understanding of the relationship between amount of genetic diversity expression of morphological and agronomic characters and adaptation to stress environments may elucidate whether the success of landraces in less favourable areas is due to a population buffering mechanism or to a particular architecture of morpho-physiological traits, or both. This may in turn clarify whether 'pure line breeding' is the correct approach for less favourable areas (Ceccarelli et al., 1987). Pure line breeding can be successful only if genotypes with a very high degree of phenotypic plasticity are identified.

Evans (1980) pointed out that selection for adaptation may result in yield increases but may not represent selection for greater yield potential. However, it has already been shown for durum wheat (Pecetti et al., 1992) and barley that some of the material selected under unfavourable conditions is able to retain its superiority in a more favourable environment (Ceccarelli et al., 1991). For barley the proportion was about 20% of the selected genotypes, and for durum wheat about 30%. In both cases such a proportion was higher than the proportion of lines selected under favourable conditions which were also able to perform well in a less favourable environments, and this is in agreement with previous observations (Pecetti et al., 1994).

Breeding for specific adaptation is particularly important in the case of crops predominantly grown in unfavourable conditions, because unfavourable environments tend to be more different from each other than favourable environments (Ceccarelli and Grando, 1997). The specific adaptation strategy may be explored on the basis of yield response of the germplasm pool that is representative of the available genetic base tested across a representative sample of sites within the target region (Annichiarico, 2002).

# 9.3 Conversion into Density-Neutral Modern Cultivars

Inter-plant competition, i.e., the unequal sharing of growth resources due to genetic (pre-existing) or acquired differences among plants, can be quantified by the drastic increase in the coefficient of variation (CV) of individual plant yields in the crop stand (Fasoula and Fasoula, 1997). The difference between the genetically heterogeneous landraces and the genetically highly homogeneous modern cultivars means that a landrace stand involves genetic competition among plants, whereas a modern cultivar stand is devoid of genetic competition. A systematic study of the relationship between yield and competitive ability within a bread wheat cultivar (Fasoula, 1990) found a high and significant negative correlation (r = 0.94). The study demonstrated that highly competitive plants, i.e., those yielding less at the ultra-low planting density

(1 plant/m<sup>2</sup>; absence of competition), out-yielded the low competitors in mixed stands, i.e., plants yielding more at the ultra-low density. Conversely, in pure stands, the performance of highly competitive genotype plants lagged behind that of the poor competitors.

Genetic heterogeneity, such as that found in landraces, involves genetic competition among plants, in addition to acquired competition, which is also encountered in stands of single genotypic cultivars. In principle, genetic competition can be eliminated when landraces are converted into desirable homozygous lines, but the acquired competition is more difficult to control. The intensity of acquired competition increases in marginal or low input environments, which possess inherent heterogeneity in the distribution of resources. The above offers an insight into the reasons that led to the eventual replacement of landraces by pure single genotype modern cultivars in favourable environments and their persistence in marginal environments. In favourable environments, competition in the crop stands is reduced because of elimination of both the genetic (single genotype) and the acquired (ample resources) components. This results in increased crop yields, reflected in the reduced CV of individual plant yields (Fasoula and Fasoula, 1997; Tollenaar and Wu, 1999).

Yields of modern single genotype pure cultivars can be either density-dependent as in the case of maize hybrids that yield optimally under high plant densities only (Duvick, 1992), or density-neutral, i.e., remain optimal under a wide range of plant densities. With appropriate breeding methodology, it is eventually possible to convert the density-dependent into more density-neutral cultivars (Fasoula and Fasoula, 2000; Tokatlidis et al., 2001). To effectively exploit landrace diversity for breeding purposes, it is important to appreciate the significance of creating more densityneutral genotypes, particularly for marginal environments, which usually suffer from drought stress. In drought-prone environments, the use of more densityneutral cultivars allows use of lower seeding rates, limiting the damage due to drought. A subtle point is the understanding that density itself is not a stress; it simply enhances the existing differences that lead to competition in the stand. Therefore, particularly in more uniform environments, it is possible to have a high yielding dense stand with reduced inter-plant competition, as measured by the reduced CV of individual plant yields.

There is a lot of evidence that mixtures can be valuable, especially when combining various disease resistances of the components. A word of caution is, however, presented as to the interpretation behind the phenomenon. In favourable environments, the evidence of yield stability due to heterogeneity is often counter-balanced by experiments and theory, indicating that increasingly homogeneous cultivars have higher yield potential across sites and years. A case in point is the well-known, superior stability of performance of the genetically homogeneous modern single-cross maize hybrids over the older genetically heterogeneous double-cross hybrids (Duvick, 1992), because the adverse effects of genetic competition are restricted by genetic uniformity. This hints to the nonuniversality of the superiority of populational (Allard and Bradshaw, 1964) buffering. However, the requirement for superior individual buffering is that genes conferring tolerance to the biotic and abiotic stresses are being gradually incorporated into a few or a single individual(s). Further, the cause of the superior stability of certain mixtures has been indicated (Fasoula and Fasoula, 1997) to mainly rest in the reduced stability of the individual components (quantified by the higher CV of single plant yields in pure stands), appearing as reduced under-compensation in the mixtures, mimicking the effects of true over-compensation (pseudoovercompensation).

# 9.4 Seed Degradation – Cultivar Degeneration

An additional detrimental effect of the negative correlation between yield and competitive ability relates to the observed landrace or cultivar "seed degradation". Zeven (1999) provides an interesting array of evidence about what he calls the often inexplicable seed replacement by traditional farmers. A traditional practice to combat seed degradation has been the periodic seed replacement of farmers' own seed with seed from elsewhere. Zeven (2000) reports a widely existing belief that the home-grown cultivar degenerates after several generations of re-sowing. He further states (Zeven, 2000) that most farmers do not actually perform traditional maintenance breeding; as they and their ancestors probably have experienced that traditional maintenance breeding does not result in a better crop. Apparently, "farmers must have thought that seed replacement was a better method to maintain the yielding capacity of their crops".

A proposed explanation for the practice of "inexplicable" seed replacement and the avoidance of traditional maintenance breeding by farmers relates to the consequences of the existing, but mostly unsuspected, negative correlation between yield and competitive ability. The problem of cultivar or landrace degeneration (Fasoula, 1990) can be addressed by applying the concept of non-stop selection (Fasoula and Fasoula, 2000) for superior lines at ultra-low plant densities. The outcomes of non-stop selection exceed that of conventional maintenance breeding (Fasoula and Boerma, 2007). Experimental data of honeycomb selection within breeder's seed of the old barley cultivar Athenais and the old durum wheat cultivar Kyperounda, both of local landrace origin, demonstrated the existence of useful adaptive variation persisting within homozygote lines, as well as the potentialities of non-stop selection (Fasoula, 2004).

Adaptive variation is genome-monitored, de novo and heritable across generations. It is directional and constantly released by the sensory mechanisms of the genome in response to environmental stimuli (McClintock, 1984; Rasmusson and Philips, 1997). This epigenetic variation stems from the interaction between genotype and environment (Goldberg et al., 2007) and allows profitable exploitation of limited resources and continual incorporation of gene variants for resistance to changing biotic and abiotic stresses. Continuous exploitation of adaptive variation is synonymous with the continuous genetic upgrading of landraces and cultivars.

Because of the reported negative correlation between yield and competitive ability, integration of yield and stability genes into fewer, improved genotypes is more efficient when the unit of evaluation and selection becomes the individual plant grown at ultra-low planting densities. When this negative correlation is considered at the level of the individual plant, it means that a plant possessing genes for high yield potential will also possess genes for low competitive ability. At the level of the variety crop stand, the negative correlation means that the greater the inter-plant competition in the stand, quantified by the yield CV of individual plants, the greater the crop yield reduction. At the level of selection and landrace/variety maintenance, it means that high competitors are selected at the expense of higher yielding genotypes. As a result, the variety eventually degenerates (Fasoula, 1990). This phenomenon offers a novel explanation for the previously reported practice of landrace seed replacement amongst traditional farmers and their avoidance of traditional maintenance breeding.

In summary, single gene traits are generally easily transferred from landrace germplasm to modern cultivars, but most of the desirable traits characteristic of landraces are complex and difficult to express in different genetic backgrounds. Maintaining these characteristics in heterogeneous landraces is also problematic. Breeding, selection and deployment methods appropriate to these objectives should be used rather than those used for high input intensive agriculture plant breeding.

## 10 Participatory Breeding

Participative approaches to agricultural research and development are now extensively used throughout the world to help define and address the practical research needs of farmers. They have proved useful in solving practical problems in complex and diverse farming systems characteristic of organic farming and low input systems. During the last few years participatory research divided into several different topics and in the field of plant breeding there are already many very different strategies like Participatory Varietal Selection, true or complete Participatory Plant Breeding (Witcombe et al., 1996) and an intermediate approach, Efficient Participatory Breeding (Morris and Bellon, 2004). The essential advantages of participatory plant breeding over conventional plant breeding involve: better targeting of local environmental conditions, better definition of selection criteria important to the endusers, faster and greater adoption of improved cultivars by the farmer, and increase or maintenance of genetic variability. Participatory plant breeding also gives voice to farmers and elevates local knowledge to the status of science (Ceccarelli and Grando, 1997).

Very effective participatory plant breeding projects in cereals are active all over the world. In Europe in particular, a few participatory plant breeding projects are running with success either in cross-pollinated and self-pollinate cereals. In Portugal, the VASO (Vale do Sousa-Sousa Valley) project, running since 1984, is a maize Participarory Plant Breeding project developed to cover the needs of small maize farmers, with scarce land resources, in poly-cropping systems for human uses, particularly bread production (Moreira, 2006). Local germplasm is used, adapted to the local conditions over centuries of cultivation, quality being the first priority over quantity. This Participarory Plant Breeding project concerns mainly flint-type open-pollinated landraces with quality for the production of the traditional maize bread called 'broa'. This quality depends on traits not present in the available commercial hybrid cultivars (Brites et al., 2008). 'Broa' production still plays an important economic and social role in Central and Northern Portuguese rural communities. Selection takes place at the farmers' field by the farmer in close collaboration with the breeder.

The genetic diversity evolution through this participatory breeding project was evaluated using molecular markers and it was concluded that though interesting phenotypic improvements were achieved, the level of genetic variability was not significantly influenced and diversity was maintained (Vaz Patto et al., 2007). Several maize open-pollinated landraces were selected within this project with the joint collaboration of the breeder and the farmers. In Sweden, the Allkorn project, running since 1995, is a cereal Participarory Plant Breeding project for organic farming (Larsson, 2006). The goal of the project is to identify and find interesting cereal cultivars for organic farming in Sweden. Cultivars should be of high quality for human nutrition and be well adapted to local soils and climate. They should have good weed-suppressive traits as well as tolerance to diseases and pests. The farmer can select the best cultivar, from his point of view, from several quality cultivars that he has chosen from the project and tested in his own fields. The main aim of this breeding is not to conserve the cultivars but to develop them for the future use in organic farming all over the country. Cultivars of nearly all species of cereals which have historically been used in Sweden are tested in this project, namely Triticum monococcum, T. dicoccum, T. spelta, winter and spring wheat T. aestivum, winter and spring rye Secale cereale, hulless barley, spring barley, oats, and black oats. These cultivars have a broad diversity and cover primitive cultivars, older landrace cultivars and early Swedish breeding cultivars from 1900 to 1950, since in this

early breeding period one of the parents of the cultivars was often a local landrace. The cultivars are selected each year for better adaptation to organic conditions. The idea is to form province groups of farmers that could help each other with seed supply, seed cleaning, shelling of spelt wheat, milling and selling. Regional production groups are formed for the local market with help of local mills and local bakers and the consumer can then find heritage cultivars from each region, country or province. The goal of this type of participatory research is empowering the farmers: supporting the formation of groups capable of assessing their own needs and addressing them either directly or through demands on research organisations.

In Hungary several Participatory Varietal Selection projects are running both in on farm cereal landrace conservation and organic farming research. The on-farm conservation Participatory Varietal Selection strategy is mainly connected with the regeneration of traditional Hungarian landraces and old cultivars which were maintained only in ex situ collections over the last 20 years. Several organic farmers are involved in such projects and their main interest is related to the traditional use of special local food and heritage farming in relation to agro-tourism. In such Participatory Varietal Selection programmes, the Hungarian National Genebanks provide accessions to several farmers participating in the in situ conservation project, and the locations for long-term maintenance are chosen by all participants in order to try to find the optimal place (environmental condition) and landrace to maintain, considering the social background of the region too (Holly, 2000). In recent years new projects have been initiated in the field of Participatory Varietal Selection in Hungary. One of the biggest ones, entitled "Selection of suitable cultivars for organic farming" was carried out in a real Participatory Varietal Selection system in which most of the important stakeholders - the breeders, seed producers, farmers and end users - were involved in cultivar selection for several field crops, including wheat, barley, sunflower, pea, etc. (Kovács, 2006a). In this project breeding institutions provided both landraces and old and modern cultivars and farmers and end user made the cultivar evaluation independently of the providers' opinion in each locality taking into consideration the end users priorities (Kovács et al., 2006). A similar research programme is running for underutilised cereal species, such as einkorn, emmer and macha wheat in the country regions where such species were traditionally used. In this case the re-introduction of the forgotten species into modern agriculture it is important to exploit the genetic diversity which still exists in such cereals (Kovács, 2008).

A participatory breeding programme in tetraploid wheat was initiated in 2001 at INRA-Montpellier, France, based on a demand for organic pasta by the industry (Desclaux, 2005). The quality of the wheat cultivars being produced under organic conditions did not meet the requirements of the processing industry. To identify the main causes, a multidisciplinary public research team, bringing together plant breeders, soil scientists, ecologists, agronomists and sociologists was assembled to work in close collaboration with the farmers and end-users. The two main French territories involved were Camargue and Pays Cathare (Desclaux et al., 2002). The lack of cultivars adapted to low nitrogen conditions became apparent rapidly in such organic crop systems. The main aim of the on-farm and participatory breeding is to take on-board farmers preferences and to better target local environmental conditions by increasing and managing the genetic variability.

Old durum wheat cultivars, segregating or advanced pure lines and populations resulting from crosses between durum wheat and emmer or wild species were provided to the farmers by the breeders. Selection and evaluation are done in close collaboration and results are discussed between all the participants. Information is feed back and can lead to re-examination of the objectives of the breeding scheme (Desclaux, 2005). This French project is neither a farmer-led nor a formalled project but it is led by both professionals and researchers and requires farmer's critical participation right from the first steps of the breeding scheme, i.e. it is collegiate and decentralised (Desclaux, 2005).

In the UK, a project for developing appropriate participatory methodologies for cereal seed production and cultivar selection under organic conditions has been established (Clarke et al., 2006). This project aimed to overcome the deficiencies of the UK official Recommended List of Cereals for organic farmers. In winter wheat comparative trials using cultivars and mixtures of cultivars were established at 19 UK farms the 2003/04 and 2004/05 seasons (Clarke et al., 2006; Jones et al., 2006). A collegiate participatory method was used to balance statistical rigour with farmer's objectives of managing a whole farm system. Participating farmers sowed the seed in large marked plots using their standard methodology within a field containing wheat. Researchers gained information from each of the farmers about their farming system, field and trial. These experiments showed the large variability of organic systems in the UK and the difficulty of selecting a single cultivar suitable for them all. This project also helped to develop a small core of trained farmers and researches which can be exploited in further participatory projects.

The diversity available within landraces is useful for breeding purposes in at least four different ways (Ceccarelli et al., 1987), including the release of the highest yielding lines as pure line cultivars, the utilisation of superior lines as parents, the evaluation of multilines built with a variable number of pure lines, and the identification of lines showing extreme expression of specific attributes.

Participatory plant breeding and variety selection has proven more successful than the approach used in high input breeding programmes for landrace improvement in stress-prone environments where sustainable approaches are a high priority. Despite being more complex to carry out, it not only delivers improved germplasm, but also aids uptake and communication between farmers, researchers and advisors for the benefit of all.

#### 11 Legal Issues

Landraces as an important genetic resource have been included in international treaties and national decrees that protect and enhance their use in their local environment. The objectives of the convention on biological diversity are the conservation of biological diversity, the sustainable use of its components and the fair and equitable sharing of the benefits arising out of the utilisation of genetic resources. The objectives of the International treaty on plant genetic resources for food and agriculture (http://www.planttreaty.org) are the conservation and sustainable use of plant genetic resources and the fair and equitable sharing of benefits derived from their use, in harmony with the Convention on Biological Diversity, for sustainable agriculture and food security. The treaty promotes or supports, as appropriate, farmers and local communities' efforts to

manage and conserve on-farm their genetic resources for Food and Agriculture.

In Europe the marketing of seeds of landraces and conservation of cultivars is ruled by the strict seed trade and cultivar protection laws. Cultivars have to meet the Distinctiveness, Uniformity and Stability standards to be registered legally. Agricultural crops have to meet the Value of Cultivation and Use criteria and to be listed on the "European common catalogue of cultivars of agricultural plant species" to be tradable within Europe. Combined with the extension of the intellectual property rights in the UPOV (French: Union internationale pour la protection des obtentions végétales) 91 act, the maintenance of agro-biodiversity onfarm and the conservation of local cultivars are generally threatened. According to the Biodiversity Action Plan for agriculture presented by the European Commission in 2001 "the conservation and improvement of in situ/on farm plant genetic resources also depends on the effective possibility of sustainable uses and on legislation which makes it possible to market diversified genetic materials".

The directive of conservation cultivars is still limiting the diversity of seed. The most recent draft of a directive of conservation cultivars has the same goals as the UPOV standards: Uniformity, stability and distinctness, but less documentation is needed to fulfil the requirements. Landraces are never uniform, neither are they stable. Therefore, landraces will not be allowed according to the directive of conservation cultivars. The directive of conservation varieties therefore applies for old out-dated approved cultivars, and will not contribute to the biodiversity of arable land (SANCO, 2006). An international commission has written a manifest on the future of seed in order to strengthen the movement toward sustainable agriculture, food sovereignty, biodiversity and agricultural diversity, to help defend the rights of farmers to save, share, use and improve seeds and enhance our collective capacity to adapt to the hazards and uncertainties of environmental and economic change (Shiva, 2007). The most recent European Union Commission Directive 2008/62/EC published on 20th June 2008 does provide for certain derogations for acceptance of agricultural landracs and varieties which are naturally adapted to the local and regional conditions and threatened by genetic erosion and for marketing of seed of those landraces and varieties.

Clearly legislation was designed primarily to protect trade and return royalty income to modern plant breeders with expensive programmes to fund. As the desirability of using landraces becomes more apparent to achieve greater sustainability, legislation changes are being made to facilitate this trade too. However, more changes are needed to promote the exploitation of diversity in landraces and encourage their use.

## 12 Conclusions

The position of cereal landraces as valuable genetic material in contemporary agriculture is gaining renewed importance. A lot of recent research effort has gone into collecting, organising, studying and analysing them with a primary goal being to incorporate their positive qualities in new cultivars or populations for a more sustainable agricultural production, particularly in response to recent climate changes. Positive attributes include landraces being a source of novel resistance genes or combination of genes with a good deployment strategy. Particularly important is the fact that resistance found in cereal landraces has been durable in many instances, in contrast to the gene-forgene type of resistance that is usually encountered in modern cultivars. However, the durability may be associated with and dependent on deployment in heterogeneous populations.

A major part of this valuable landrace diversity is conserved in the world's gene banks network and should be exploited systematically for traits such as quality and specific adaptation to stress environments. The available genetic variation in adaptive responses to soil and climatic conditions conserved in landraces is little understood, known and even less used. More uniform and user-friendly documentation about collection and characterisation of landraces, either morphologically or with molecular tools, is needed to access this variation more effectively. Gene banks should aim at adopting a common concept of landraces and plan special inventories for them. The level of diversity should be monitored during their conservation so that the original level of variation is maintained. More studies are needed in order to investigate if their long-term maintenance by farmers resulted in increasing genetic variation.

New high-throughput genotyping platforms and phenotyping data in common databases will enable powerful association genetics approaches to be utilised for improvement and utilisation of landrace resources. Knowledge of linkage disequilibria in landraces compared with elite germplasm will help focus breeding for stress-adapted cultivars or populations. With the current rapid advancements in high-throughput molecular genotyping technologies, it is becoming increasingly clear that the limiting factor in applying those powerful technologies to molecular breeding programmes is no more the capacity of genotyping, but the potential for accurate or precision phenotyping. Particularly important and challenging is phenotyping for the so-called quantitative traits.

The renewed focus on cereal landraces for breeding purposes is also a response to some negative consequences of modern agriculture and conventional breeding, such as the liberal use of high inputs, the loss of genetic diversity, and the stagnation of yields in less favourable areas. To deliver this, participatory plant breeding and variety selection practices have emerged as a powerful way to merge breeders' knowledge and farmers' selection criteria, emphasizing decentralised selection in the target environments with the active participation of local farmers. Location-specific adaptation of these diverse landraces will be important for further selection at the farmer level. New strategies are emerging to produce "modern landraces" based on multiple cross populations of einkorn, emmer and bread wheat in combination with on farm site-specific selection to obtain highly adaptable populations for local and regional production.

The reported practice of the "inexplicable seed replacement" by traditional farmers is connected to the gradual seed degradation during landrace maintenance, brought about by the established negative correlation between breeding and competitive ability. Further enhancement of productivity and stability is achieved through practicing "non-stop selection" within landraces across the marginal production environments, to exploit the constantly released by the genome useful adaptive variation. The procedure and results of non-stop selection exceed those of conventional maintenance breeding, combating seed degradation and resulting to a constant cultivar upgrade. An additional essential breeding consideration is the creation of density-neutral landrace germplasm, since its use, particularly in marginal environments, permits lower seeding rates, limiting the damage due to drought.

The issues of conservation and sustainable use of landraces have been so far included in international convention treaties and directives which need to be implementated on European and national levels. In this review we have highlighted the value of landraces as resources for the future sustainability of cereal crop production, the methods to enhance their genetic makeup and avoid seed degradation, and emphasised the level of coordination and resourcing needed to realise their great potential.

Acknowledgements We thank the European Union for funding the COST Action 860 "Sustainable variety development for lowinput and organic agriculture" (2004–2008) through which this review was initiated.

## References

- Abdellaoui R., M'Hamed H.C., Naceur M.B., Bettaïeb-Kaab L., Hamida J.B. (2007) Morpho-physiological and molecular characterization of some Tunisian barley ecotypes, Asian J. Plant Sci. 6, 261–268.
- Abo-Elenin R.A., Heakal M.S., Gomaa A.S., Moseman J.G. (1981) Studies on salt tolerance in barley and wheat. Source of tolerance in barley germplasm, Barley Genet. IV, 402–409.
- Acevedo E., Fereres E. (1993) Resistance to abiotic stresses. in: Hayward M.D., Osemark N.O., Romagosa I. (Eds.), Plant Breeding and Prospects, Chapman & Hall, London, UK, pp. 406–421.
- Adom K.K., Liu R.H. (2002) Antioxidant activity of grains, J. Agr. Food Chem. 50, 6182–6187.
- Afanasenko O.S., Terentyeva I.A., Makarova I.N. (2000) Landraces from Peru – new sources of resistance to net blotch of barley, Proceedings of the 8th International Barley Genetics Symposium, Adelaide, October 22–27 2000, Vol. II, pp. 71–72, Adelaide University, Australia.
- Akar T., Özgen M. (2007) Genetic Diversity in Turkish Durum Wheat Landraces, in: Wheat Production in Stressed Environments, Springer, Netherlands, 12, pp. 753–760.
- Akar T., Francia E., Tondelli A., Rlzza F., Stanea A.M., Pecchioni N. (2009) Marker-assisted characterization of highly Frost Tolerant Barley (*Hordeum vulgare* L.) Genotypes, Plant Breeding 128, 381–386.
- Akhkha A., Clarke D.D., Dominy P.J. (2003) Relative tolerances of wild and cultivated barley to infection by *Blumeria graminis* f.sp. *hordei* (Syn. *Erysiphe graminis* f.sp. *hordei*). II – the effects of infection on photosynthesis and respiration, Physiol. Mol. Plant. Path. 62, 347–354.
- Alamerew S., Chebotar S., Huang X., Röder M., Börner A. (2004) Genetic diversity in Ethiopian hexaploid and tetraploid wheat germplasm assessed by microsatellite markers, Genet. Resour. Crop Ev. 51, 559–567.

- Al Khanjari S., Hammer K., Buerkert A., Röder M.S. (2007) Molecular diversity of Omani wheat revealed by microsatellites: II. Hexaploid landraces, Genet. Resour. Crop Ev. 54, 1407–1417.
- Alemayehu F., Parlevliet J.E. (1997) Variation between and within barley landraces, Euphytica 94, 183–189.
- Allard R.W., Bradshaw A.D. (1964) Implications of genotypeenvironmental interactions in applied plant breeding, Crop Sci. 4, 503–508.
- Almekinders C.J.M., Louwaars N.P., de Bruijn G.H. (1994) Local seed systems and their importance for an improved seed supply in developing countries, Euphytica 78, 207–216.
- Anderson J.B., Torp J. (1986) Quantitative analysis of the early powdery mildew infection stages on resistant barley genotypes, J. Phytopathol. 115, 173–185.
- Andrade V., dos Santos T.M.M., Afonso Morales D., Costa G., Pinheiro de Carvalho M.A.A. (2007) Evaluation of wheat germplasm at the Madeira and Canary Archipelagos, using a single molecular marker. A rapid screening method for durum wheat identification, Cereal Res. Commun. 35, 1397–1404.
- Anikster Y., Noy-Meir I. (1991) The wild-wheat field laboratory at Ammiad, Israel J. Bot. 40, 351–362.
- Anikster Y., Wahl I. (1979) Coevolution of rust fungi on Gramineae and Liliaceae and their hosts, Ann. Rev. Phytopathol. 117, 367–430.
- Anikster Y., Manisterski J., Long D.L., Leonard K.J. (2005) Leaf rust and stem rust resistance in *Triticum dicoccoides* populations in Israel, Plant Dis. 89, 55–62.
- Aniol A., Madej L. (1996) Genetic variation for aluminum tolerance in rye, Vortr. Pflanzenz, Chtg. 35, 201–211.
- Annichiarico P. (2002) Defining adaptation strategies and yield stability targets in breeding programmes, in: Kang M.S. (Ed.), Quantitative genetics, genomics and plant breeding, CABI, Wallingford, UK, pp. 165–183.
- Annicchiarico P., Pecetti L. (1993) Contribution of some agronomic traits to durum wheat performance in a dry Mediterranean region of Northern Syria, Agronomie 13, 25–34.
- Annichiarico P., Pecetti L. (1998) Yield vs. morphophysiological trait-based criteria for selection of durum wheat in a semiarid Mediterranean region (northern Syria), Field Crop. Res. 59, 163–173.
- Anon. (1943) Annual Report, Wheat Sub-Station, Gurdaspur, India, pp. 1–15.
- Arraiano L.S., Brown J.K.M. (2006) Identification of isolatespecific and partial resistance to septoria tritici blotch in 238 European wheat cultivars and breeding lines, Plant Pathol. 55, 726–738.
- Asher M.J.C., Thomas C.E. (1983) The genetical control of incomplete forms of resistance to *Erysiphe graminis* in spring barley, Ann. Appl. Biol. 103, 149–156.
- Asher M.J.C., Thomas C.E. (1984) Components of partial resistance to *Erysiphe graminis* in spring barley, Plant Pathol. 33, 123–130.
- Asher M.J.C., Thomas C.E. (1987) The inheritance of mechanisms of partial resistance to *Erysiphe graminis* in spring barley, Plant Pathol. 36, 66–72.
- Assefa A., Labuschagne M.T. (2004) Phenotypic variation in barley (*Hordeum vulgare* L.) landraces from north Shewa in Ethiopia, Biodivers. Conserv. 13, 1441–1451.

- Assefa A., Labuschagne M.T., Viljoen C.D. (2007) AFLP analysis of genetic relationships between barley (*Hordeum vul*gare L.) landraces from north Shewa in Ethiopia, Conserv. Genet. 8, 273–280.
- Attene G., Veronesi F. (1991) Observations on a Sardinian Population of Six Rowed Barley (*Hordeum vulgare* L.) (Italian), Riv. Agron. 25, 54–56.
- Autrique E., Nachit M.M., Monneveux P., Tanksley S.D., Sorrels M.E. (1996) Genetic diversity in durum wheat based on RFLPs, morpho-physiological traits, and coefficient of parentage, Crop Sci. 36, 735–742.
- Backes G., Hatz B., Jahoor A., Fischbeck G. (2003) RFLP diversity within and between major groups of barley in Europe, Plant Breeding 122, 291–299.
- Baigulova G.K., Pitonya A.A. (1979) *Helminthosporium* on barley, Uzbekiston Biologija Zurnali 4, 55–57.
- Bálint A.F., Kovacs G., Sutka J. (2003) Comparative studies on the seedling copper tolerance of various hexaploid wheat varieties and of spelt in soil with a high copper content and in hydroponic culture, Acta Agr. Hungarica 51, 199–203.
- Banttari E.E., Anderson W.H., Rasmusson D.C. (1975) *Helminthosporium* headblight resistance in six-row spring barleys, Plant Dis. Rep. 59, 274–277.
- Baon J.B., Smith S.E., Alston A.M. (1993) Mycorrhizal response of barley cultivars differing in P efficiency, Plant Soil 157, 97–105.
- Baresel J.P., Reents H.J., Zimmermann G. (2005) Field evaluation criteria for nitrogen uptake and nitrogen use efficiency, Proceedings of the Workshop on Organic Breeding Strategies and the Use of Molecular Markers Driebergen, The Netherlands, 17–19 January 2005, organised by COST860 SUS-VAR Working Group 1 and ECO-PB; in collaboration with COST 851, Working Group 3.
- Baresel J.P., Zimmermann G., Reents H.J. (2008) Effects of genotype and environment on N uptake and N partitioning in organically grown winter wheat (*Triticum aestivum*. L.) in Germany, Euphytica (in press), DOI 10.1007/s10681-008-9718-1.
- Ben Amer I.M., Börner A., Röder M.S. (2001) Detection of genetic diversity in Libyan wheat genotypes using wheat microsatellite markers, Genet. Resour. Crop Ev. 48, 579–585.
- Bertholdsson N.O., Stoy V. (1995) Yields of Dry Matter and Nitrogen in Highly Diverging Genotypes of Winter Wheat in Relation to N-uptake and N-Utilisation, J. Agron. Crop Sci. 175, 285–295.
- Bingham I.J., Newton A.C., (2009) Crop tolerance of foliar pathogens: possible mechanisms and potential for exploitation, in: Walters D. (Ed.), Non-fungicidal approaches to disease control in crops, Blackwell Publishing Oxford, UK, pp. 142–161.
- Bisht I.S., Mithal S.K. (1991) Evaluation of barley germplasm for resistance against stripe disease, Indian Phytopathol. 44, 516–517.
- Bjørnstad A., Demissie Å., Kilian A., Kleinhofs A. (1997) The distinctness and diversity of Ethiopian barleys, Theor. Appl. Genet. 94, 514–521.
- Bjørnstad Å., Grønnerød S., Mac Key J., Tekauz A., Crossa J., Martens H. (2004) Resistance to barley scald (*Rhynchosporium secalis*) in the Ethiopian donor lines 'Steudelli' and 'Jet', analyzed by partial least squares regression and interval mapping, Hereditas 141, 166–179.

- Blum A. (1988) Plant Breeding for Stress Environments. CRC Press, Boca Raton, Florida.
- Bolan N.S. (1991) A critical review on the role of mycorrhizal fungi in uptake of phosphorus by plants, Plant Soil 134, 189–207.
- Bonman J.M., Bockelman H.E., Jin Y., Hijmans R.J., Gironella A. (2007) Geographic distribution of stem rust resistance in wheat landraces, Crop Sci. 47, 1955–1963.
- Bothmer R., Hintum T., Knuepffer H., Sato K. (2003) Diversity in barley (*Hordeum vulgare*), Developments in Plant Genetics and Breeding, Vol. 7, Elsevier Science B.V.
- Brandolini A. (1969) European races of maize, in: Proceedings of the 24th Corn and Sorghum Research Conference, pp. 34–48.
- Brandolini A., Brandolini A. (2001) Classification of Italian maize (*Zea mays* L.) germplasm, Plant Genet. Res. Newsl. 126, 1–11.
- Bregitzer P., Mornhinweg D.W., Obert D.E., Windes J. (2008) Registration of 'RWA 1758' Russian Wheat Aphid-Resistant Spring Barley, J. Plant Reg. 2, 5–9.
- Brites C., Trigo M.J., Santos C., Collar C., Rosell C.M. (2008) Maize-based gluten-free bread: Influence of processing parameters on sensory and instrumental quality, Food Bioprocess Technol. 226, 1205–1212.
- Bryngelsson S., Mannerstedt-Fogelfors B., Kamal-Eldin A., Andersson R., Dimberg L.H. (2002) Lipids and antioxidants in groats and hulls of Swedish oats (*Avena sativa* L), J. Sci. Food Agr. 82, 606–614.
- Buchannon K.W., McDonald W.C. (1965) Sources of resistance in barley to *Pyrenophora teres*, Can. J. Plant Sci. 45, 189–193.
- Buckler IV E.S., Thornsberry J.M. (2002) Plant molecular diversity and applications to genomics, Curr. Opin. Plant Biol. 5, 107–111.
- Caballero L., Martín L.M., Alvarez J.B. (2001) Allelic variation of the HMW glutenin subunits in Spanish accessions of spelt wheat (*Triticum aestivum* ssp. *spelta* L. em. Thell.), Theor. Appl. Genet. 103, 124–128.
- Caldwell K.S., Russell J., Langridge P., Powell W. (2006) Extreme population-dependent linkage disequilibrium detected in an inbreeding plant species, *Hordeum vulgare*, Genetics 172, 557–567.
- Camacho Villa T.C., Maxted N., Scholten M.A., Ford-Lloyd B.V. (2005) Defining and identifying crop landraces, Plant Genet. Res. 3, 373–384.
- Campos H., Cooper M., Habben J.E., Edmeades G.O., Schusser J.R. (2004) Improving drought tolerance in maize: a view from industry, Field Crop. Res. 90, 19–34.
- Canevara M.G., Romani M., Corbellini M., Perenzin M., Borghi B. (1994) Evolutionary Trends in morphological, physiological, agronomical and qualitative traits of *Triticum aestivum* L. cultivars bred in Italy since 1900, Eur. J. Agron. 3, 175–185.
- Cartwright B., Zarcinas B.A., Mayfield A.H. (1984) Toxic concentrations of boron in a red-brown earth at Gladstone. South Australia, Aust. J. Soil Res. 22, 261–272.
- Carver T.L.W. (1986) Histology of infection by *Erysiphe grami*nis f. sp. hordei in spring barley lines with various levels of partial resistance, Plant Pathol. 35, 232–240.
- Carver B.F., Ownby J.D. (1995) Acid soil tolerance in wheat, Adv. Agron. 54, 117–173.

- CBD (1993) Convention on Biological Diversity, http://www.cbd.int/convention/convention.shtml.
- Ceccarelli S. (1987) Yield potential and drought tolerance of segregating populations of barley in contrasting environments, Euphytica 36, 265–273.
- Ceccarelli S. (1994) Specific adaptation and breeding for marginal conditions, Euphytica 77, 205–219.
- Ceccarelli S. (1996) Adaptation to low high-input cultivation, Euphytica 92, 203–214.
- Ceccarelli S., Grando S. (1997) Increasing the efficiency of breeding through farmer participation, in: Ethics and Equity in Conservation and use of Genetics Resources for Sustainable Food Security, Proceeding of a workshop to develop guidelines for the CGIAR, 21–25, April 1997, Foz de Iguacu, Brazil, IPGRI, Rome, pp. 116–121.
- Ceccarelli S., Acevedo E., Grando S. (1991) Breeding for yield stability in unpredictable environments: single traits, interaction between traits and architecture of genotypes, Euphytica 56, 169–185.
- Ceccarelli S., Grando S., Van Leur J.A.G. (1987) Genetic diversity in barley landraces from Syria and Jordan, Euphytica 36, 389–405.
- Ceccarelli S., Piano E., Arcioni S. (1976) Evaluation of selected F4 and F5 families in a pedigree programme in barley (*Hordeum vulgare* L.), Abstracts of papers presented at the XX Annual Meeting, Italian Society for Agricultural Genetics, Papers on miscellaneous crop plants 30, 91.
- Chantret N., Salse J., Sabot F., Rahman S., Bellec A., Laubin B., Dubois I., Dossat C., Sourdille P., Joudrier P., Gautier M.-F., Cattolico L., Beckert M., Aubourg S., Weissenbach J., Caboche M., Bernard M., Leroy P., Chalhoub B. (2005) Molecular basis of evolutionary events that shaped the hardness locus in diploid and polyploid wheat species (*Triticum* and *Aegilops*), Plant Cell 17, 1033–1045.
- Chartrain L., Brading P.A., Brown J.K.M. (2005) Presence of the *Stb6* gene for resistance to *Septoria tritici* blotch (*Mycosphaerella graminicola*) in cultivars used in wheat-breeding programmes worldwide, Plant Pathol. 54, 134–143.
- Cherdouh A., Khelifi D., Carrillo J.M., Nieto-Taladriz M.T. (2005) The high and low molecular weight glutenin subunit polymorphism of Algerian durum wheat landraces and old cultivars, Plant Breeding 124, 338–342.
- Clarke S.M., Hinchsiffe K.E., Haigh Z., Jones H., Pearce B., Wolfe M.S. Thomas J. (2006) A participatory approach to variety trials for organic systems, in: Proceedings of the Joint Organic Congress, Odense, Denmark, May 30–31, 2006.
- Cleveland D.A., Soleri D., Smith S.E. (1999) Farmer plant breeding from biological perspective: Implications for collaborative plant breeding, CIMMYT Economics Working Paper 99–10, CIMMYT, Mexico, DF, pp. 1–27.
- Clifford B.C. (1985) Barley leaf rust, The cereal rusts. Volume II, Diseases, distribution, epidemiology, and control, in: Roelfs A.P., Bushnell W.R. (Eds.), pp. 173–205.
- Coffman F.A. (1977) Oat history, identification and classification, USDA-ARS, Tech. Bull. 1516, Washington, DC.
- Comadran J., Russell J.R., van Eeuwijk F.A., Ceccarelli E.S., Grando S., Baum M., Stanca A.M., Pecchioni N., Mastrangelo A.M., Akar T., Al-Yassin A., Benbelkacem A., Choumane W., Ouabbou H., Dahan R., Bort J., Araus J.-L., Pswarayi A., Romagosa I., Hackett C.A., Thomas W.T.B.

(2007) Mapping adaptation of barley to droughted environments, Euphytica 161, 35–45.

- Corazza L., Balmas V., Chilosi G., Nalli R. (1990) Evaluation of resistance of wheat and barley varieties to attack by pathogenic species of *Fusarium*, Sement Elette 36, 25–28.
- Cosic T., Poljak M., Custic M., Rengel Z. (1994) Aluminium tolerance of durum wheat germplasm, Euphytica 78, 239–243.
- Cox M.C., Qualset C.O., Rains D.W. (1985) Genetic variation for nitrogen assimilation and translocation in wheat. I. Dry matter and nitrogen accumulation, Crop Sci. 25, 430–435.
- Czembor J.H. (2002) Resistance to powdery mildew in selections from Moroccan barley landraces, Euphytica 125, 397–409.
- D'Amato F. (1989) The progress of Italian wheat production in the first half of the 20th century: contribution of breeders, Agr. Med. 119, 157–174.
- Damania A.B., Porceddu E. (1981) Screening YAR barley for disease resistance, Plant Genet. Res. Newsl. 48, 2–3.
- Davies B.E. (1994) Soil chemistry and bioavailability with special reference to trace elements, Plant and chemical elements, in: Farago M.E. (Ed.), VCH, Weinheim, pp. 2–30.
- de Freitas L.R.A., Ganança J.F.T., dos santos T.M.M, Pinheiro de Carvalho M.Â.A., Motto M., Clemente Vieira, M.R. (2005) The use of seed proteins, zein, in the evaluation of Madeira maize germplasm, Maydica 50, 105–112.
- De Sousa C.N.A. (1998) Classification of Brazilian wheat cultivars for aluminium toxicity in acid soils, Plant Breeding 117, 217–221.
- De Vita P., Nicosia O.L.D., Nigro F., Platani C., Riefolo C., Fonzo N.D., Cattivelli L. (2007) Breeding progress in morpho-physiological, agronomical and qualitative traits of durum wheat cultivars released in Italy during the 20th century, Eur. J. Agron. 26, 39–53.
- Del Pozo-Insfran D., Brenes C.H., Saldivar S.O.S., Talcott S.T. (2006) Polyphenolic and antioxidant content of white and blue corn (*Zea mays* L.) products, Food Res. Int. 39, 696–703.
- Demissie A., Bjørnstad Å. (1997) Geographical, altitude and agro-ecological differentiation of isozyme and hordein genotypes of landrace barleys from Ethiopia: Implications to germplasm conservation, Genet. Resour. Crop Ev. 44, 43–55.
- Demissie A., Bjørnstad Å., Kleinhofs A. (1998) Restriction fragment length polymorphisms in landrace barleys from Ethiopia in relation to geographic, altitude, and agroecological factors, Crop Sci. 38, 237–243.
- Dencic S., Kastori R., Kobiljski B., Duggan B. (2000) Evaluation of grain yield and its components in wheat cultivars and landraces under near optimal and drought conditions, Euphytica 113, 43–52.
- Desclaux D. (2005) Participatory plant breeding methods for organic cereals: review and perspectives, in: Proceedings for the Eco-Pb Congress, 17–19 January 2005, Driebergen, The Netherlands, pp. 1–6.
- Desclaux D., Chiffoleau Y., Dreyfus F., Mouret J.C. (2002) Cereal cultivars innovations adapted to organic production: a new challenge, 1st International Symposium on organic seed production and plant breeding, Berlin, 21–22 November 2002.

- Dhillon B.S., Dua R.P., Brahmi P., Bisht I.S. (2004) On farm conservation of plant genetic resources for food and agriculture, Curr. Sci. 87, 557–559.
- Diederichsen A. (2008) Assessments of genetic diversity within a world collection of cultivated hexaploid oat (*Avena* sativa L.) based on qualitative morphological characters, Genet. Resour. Crop Ev. 55, 419–440.
- Dobrovolskaya O., Saleh U., Malysheva-Otto L. (2005) Rationalising germplasm collections: a case study for wheat, Theor. Appl. Genet. 111, 1322–1329.
- dos Santos T.M.M., Ganança F., Slaski J., Pinheiro de Carvalho M.Â.A. (2009) Characterization of wheat genetic resources from the *Madeira archipelago*, Genet. Resour. Crop Ev. 56, 363–370.
- dos Santos T.M.M., Pinheiro de Carvalho M.ÂA., Taylor G.J., Clemente Vieira M.R. (2005) Evaluation of the Madeiran wheat germplasm for aluminium resistance using aluminium-induced callose formation in root apices as a marker, Acta Physiol. Plant. 27, 297–302.
- Dreisigacker S., Zhang P., Warburton M.L., Skovmand B., Hoisington D., Melchinger A.E. (2005) Genetic Diversity among and within CIMMYT Wheat Landrace Accessions Investigated with SSRs and Implications for Plant Genetic Resources Management, Crop Sci. 45, 653–661.
- Du Toit F. (1987) Resistance in wheat (*Triticum aestivum*) to *Diuraphis noxia* (Hemiptera: Aphididae), Cereal Res. Commun. 15, 175–179.
- Dubey P., Mishra B. (1992) Evaluation of barley varieties against covered smut, J. Res., Birsa Agric. Univ. 4, 179–180.
- Dunaevskij A.G., Sotnikov V.V., Bibik E.V. (1989) Research for sources of resistance to the main diseases in spring barley, Selektsiya i Semenovodstvo, Kiev 67, 14–19.
- Duvick D.N. (1992) Genetic contributions to advances in yield of US maize, Maydica 37, 69–79.
- Edwards J.H., Pedersen J.F., Kingery R.C. (1990) Heritability of root characteristics affecting mineral uptake in tall fescue, in: El Bassam N., Dambroth M., Loughmann B.C. (Eds.), Genetic aspects of plant mineral nutrition, Kluwer Academic Publishers.
- Engel J.M.M., Visser L. (2003) A Guide to Effective Management of Germplasm Collections, IPGRI Handbooks for Genebanks No. 6, IPGRI, Rome, Italy.
- Evans L.T. (1980) The natural history of crop yield, Am. Sci. 68, 388–397.
- Faiad M.G.R., Wetzel M.M.V.S., Salomao A.N., Cunha R. (1996) Evaluation of fungi in seed germplasm before long term storage, Seed Sci. Technol. 24, 505–511.
- FAO (1998) The State of the World's Plant Genetic Resources for Food and Agriculture, Food and Agriculture Organization of the United Nations, Rome, 510 p.
- Fasoula D.A. (1990) Correlations between auto-, allo-, and nilcompetition and their implications in plant breeding, Euphytica 50, 57–62.
- Fasoula D.A. (2004) Accurate whole-plant phenotyping: An important component for successful marker assisted selection (MAS), Genetic variation for Plant Breeding (17th EUCARPIA General Congress), in: Vollmann J., Grausgruber H., Ruckenbauer P. (Eds.), pp. 203–206.
- Fasoula V.A. (2008) Two novel whole-plant field phenotyping equations maximize selection efficiency, Modern Variety Breeding for Present and Future Needs, in: Prohens J.,

Badenes M.L. (Eds.), Proceedings 18th EUCARPIA General Congress, pp. 361–365.

- Fasoula V.A., Boerma H.R. (2007) Intra-cultivar variation for seed weight and other agronomic traits within three elite soybean cultivars, Crop Sci. 47, 367–373.
- Fasoulas A.C., Fasoula V.A. (1995). The honeycomb selection designs, Plant Breeding Rev. 13, 87–139.
- Fasoula D.A., Fasoula V.A. (1997) Competitive ability and plant breeding, Plant Breeding Rev. 14, 89–138.
- Fasoula V.A., Fasoula D.A. (2000) Honeycomb Breeding: Principles and Applications, Plant Breeding Rev. 18, 177–250.
- Fasoula V.A., Fasoula D.A. (2002) Principles underlying genetic improvement for high and stable crop yield potential, Field Crop. Res. 75, 191–209.
- Fasoula V.A., Fasoula D.A. (2003) Partitioning Crop Yield into Genetic Components, in: Kang M.S. (Ed.), Handbook of Formulas and Software for Plant Geneticists and Breeders, Food Products Press, pp. 321–327.
- Feil B., Thiraporn R., Geisler G., Stamp P. (1990) Root traits of maize seedlings- indicators of nitrogen efficiency? Plant Soil 123, 155–159.
- Fekadu A., Parlevliet J.E. (1997) Variation between and within Ethiopian barley landraces, Euphytica 94, 183–189.
- Feldman M. (2001) Origin of cultivated wheat, in: Bonjean A.P., Angus W.J. (Eds.), The world wheat book, A history of wheat Breeding, Lavoisier Publishing, Paris, pp. 3–57.
- Feng Z.-Y., Zhang L.-L., Zhang Y.-Z., Ling H.-Q. (2006) Genetic diversity and geographical differentiation of cultivated six-rowed naked barley landraces from the Qinghai-Tibet plateau of China detected by SSR analysis, Genet. Mol. Biol. 29, 330–338.
- Ferrio J.P., Alonso N., Voltas J., Araus J.L. (2007) Grain weight changes over time in ancient cereal crops: Potential roles of climate and genetic improvement, J. Cereal Sci. 44, 323–332.
- Feuillet C., Langridge P., Waugh R. (2007) Cereal breeding takes a walk on the wild side, Trends Genet. 24, 24–32.
- Filippova G.G., Kashemirova L.A., Chiburova I.V. (1993) On seed testing in spring barley, Zashchita Rastenii Moskva 11, 29–30.
- Fischer R.A. (1981) Optimizing the use of water and nirogen through breeding of crops, Plant Soil 58, 249–278.
- Flint-Garcia S.A., Jampatong C., Darrah L.L., McMullen M.D. (2003) Quantitative trait locus analysis of stalk strength in four maize populations, Crop Sci. 43, 13–22.
- Foy C.D., Armiger W.H., Briggle L.W., Reid D.A. (1965) Differential aluminum tolerance of wheat and barley varieties in acid soils, Agron. J. 57, 413–417.
- Frankel O.H., Brown A. (1984) Plant genetic resources today: A critical appraisal, in: Genetics: New Frontiers: Proceedings of the 15th International Congress of Genetics, 1984, New Delhi.
- Frankel O.H., Brown A.H.D., Burdon J.J. (1995) The conservation of plant diversity, Cambridge: Cambridge University Press.
- Freisleben R. (1940) Die phylogenetische Bedeutung asiatischer Gersten, Züchter 12, 257–272.
- Fu Y.B., Peterson G.W., Williams D., Richards K.W., Fetch J.M. (2005) Patterns of AFLP variation in a core subset of cultivated hexaploid oat germplasm, Theor. Appl. Genet. 111, 530–539.

- Fukuyama T., Yamaji S., Nakamura H. (1998) Differentiation of virulence in *Rhynchosporium secalis* in the Hokuriku district and sources of resistance to the pathogen, Breeding Sci. 48, 23–28.
- Gahoonia T.S., Nielsen N.E. (2004a) Barley genotypes with long root hairs sustain high grain yields in low-P field, Plant Soil 262, 55–62.
- Gahoonia T.S., Nielsen N.E. (2004b) Root traits as tools for creating phosphorus efficient crop varieties, Plant Soil 260, 47–57.
- Gaike M.V. (1970) Resistance of barley cultivars to net blotch, in: Genetic basis of disease resistance of field crops, Riga, pp. 70–74.
- Galili G., Galili S., Lewinsohn E., Tadmore Y. (2002) Genetic, molecular and genomic approaches to improve value of plant foods and feeds, Crit. Rev. Plant Sci. 21, 167–204.
- Gaut B.S., Long A.D. (2003) The lowdown on linkage disequilibrium, Plant Cell 15, 1502–1506.
- Gauthier P., Gouesnard B., Dallard J., Redaelli R., Rebourg C., Charcosset A., Boyat A. (2002) RFLP diversity and relationships among traditional European maize populations, Theor. Appl. Genet. 105, 91–99.
- Gauthier P., Gouesnard B., Dallard J., Redaelli R., Rebourg C., Charcosset A., Hamza S., Hamida W.B., Rebaï A., Harrabi M. (2004) SSR-based genetic diversity assessment among Tunisian winter barley and relationship with morphological traits, Euphytica 135, 107–118.
- Geiger H.H., Heun M. (1989) Genetics of quantitative resistance to fungal diseases, Annu. Rev. Phytopathol. 27, 317–341.
- Gélinas P., McKinnon C.M. (2006) Effect of wheat variety, farming site and breadbaking on total phenolics, Int. J. Food Sci. Tech. 41, 329–332.
- Gilchrist S.L., Vivar F.H., Gonzalez C.F., Velazquez C. (1995) Selecting sources of resistance to *Cochliobolus sativus* under subtropical conditions and preliminary loss assessment, Rachis 14, 35–39.
- Goates B.J. (1986) Common bunt and dwarf bunt. Chapter 2, in: Wilcoxson R.D., Saaru E.E. (Eds.), Bunt and Smut Diseases of Wheat: Concepts and Methods of Disease Management, Mexico, D.F.: CIMMYT.
- Goldberg A.D., Allis C.D., Bernstein E. (2007) Epigenetics: A Landscape takes shape, Cell 128, 635–638.
- Goodman M.M., Paterniani E. (1969) The races of maize: III. Choices of appropriate characters for racial classification, Econ. Bot. 23, 265–273.
- Gregová E., Hermuth J., Kraic J., Dotlacil L. (1999) Protein heterogeneity in European wheat landraces and obselete cultivars, Genet. Resour. Crop Ev. 46, 521–528.
- Gregová E., Hermuth J., Kraic J., Dotlacil L. (2004) Protein heterogeneity in European wheat landraces and obsolete cultivars: Additional information, Genet. Resour. Crop Ev. 51, 569–575.
- Gregová E., Hermuth J., Kraic J., Dotlacil L. (2006) Protein heterogeneity in European wheat landraces and obsolete cultivars: Additional information II, Genet. Resour. Crop Ev. 53, 867–871.
- Grigor'ev M.F., Lukyanova M.V., Kabalkina N.A., Sidorov A.A. (1988) Sources of resistance in barley to *Fusarium* and *Helminthosporium* root rots, Sb. Nauchn. Trud. Prikl. Bot. Gen. Selektsii 122, 87–91.

- Grønnerød S., Marøy A.G., MacKey J., Tekauz A., Penner G.A., Bjørnstad A. (2002) Genetic analysis of resistance to barley scald (*Rhynchosporium secalis*) in the Ethiopian line 'Abyssinian' (CI668), Euphytica 126, 235–250.
- Gu X. (1989) Trials of several malting barley cultivars from German Federal Republic, Zuowu Pinzhong Ziyuan 1, 34–35.
- Gudu S., Maina S.M., Onkware A.O., Ombakho G., Ligeyo D.O. (2001) Screening of Kenyan maize germplasm tolerance for tolerance to low pH and aluminium for use in acid soils of Kenya, Seventh Eastern and Southern Africa Regional Maize Conference, pp. 216–221.
- Gupta P.K., Rustgi S., Kulwal P.L. (2005) Linkage disequilibrium and association studies in higher plants: Present status and future prospects, Plant Mol. Biol. 57, 461–485.
- Hammer K., Gladis T.H. (1996) Funktionen der Genbank des IPK Gatersleben bei der in situ – Erhaltung on farm, Schriften zu Genetischen Ressourcen 2, 83–89.
- Hammer K., Spahillar M. (1998) Burimet gjenetike te bimeve dhe agrobiodiversiteti, Buletini i Shkencave Bujqesore 3, 29–36.
- Hammer K., Diederichsen A., Spahillar M. (1999) Basic studies toward strategies for conservation of plant genetic resources, in: Serwinski J., Faberova I. (Eds.), Proc. technical meeting on the methodology of the FAO World Information and Early Warning System on Plant Genetic Resources, pp. 29–33, FAO, Rome, http://apps3.fao.org/wiews/Prague/Paper1.htm.
- Hamza S., Hamida W.B., Rebai A., Harrabi M. (2004) SSRbased genetic diversity assessment among Tunisian winter barley and relationship with morphological traits, Euphytica 135, 107–118.
- Hao C.Y., Zhang X.Y., Wang L.F., Dong Y.S., Shang X.W., Jia J.Z. (2006) Genetic diversity and core collection evaluations in common wheat germplasm from the Northwestern Spring Wheat Region in China, Mol. Breeding 17, 69–77.
- Hare R.A. (1997) Characterisation and inheritance of adult plant stem rust resistance in durum wheat, Crop Sci. 37, 1094–1098.
- Harlan J.R. (1975) Our vanishing genetic resources, Science 188, 618–621.
- Harlan J.R. (1977) The origins of cereal agriculture in the Old World, in: Reed C.A. (Ed.), Origins of agriculture, Moulton Publ. The Hague, Netherlands, pp. 357–383.
- Hawkes J.G., Maxted N., Floyd-Lloyd B.V. (2002) The ex situ conservation of plant genetic resources, Kluwer, The Netherlands, pp. 1–16.
- Hede A.R., Skovmand B., López-Cesati J. (2001) Acid Soils and Aluminum Toxicity, Application of Physiology in Wheat Breeding, in: Reynolds M.P., Ortiz-Monasterio J.I., McNab A. (Eds.), Mexico, D.F.: CIMMYT, pp. 172–182.
- Hetrick B.A.D., Wilson G.W.T., Cox T.S. (1993) Mycorrhizal dependence of modern wheat cultivars and ancestors – a synthesis, Can. J. Bot. 71, 512–518.
- Heuberger J.G.K., Horst W.J. (1995) Effect of root growth characteristics on nitrogen use efficiency of tropical maize (*Zea mays*) varieties, in: Jewell D.C., Waddington S.R., Ransom J.K., Pixley K.V. (Eds.), Maize research for stress environments, CIMMYT, Mexico D.F. (Mexico), pp. 44–48.
- Heun M. (1986) Quantitative differences in powdery mildew resistance among spring barley cultivars, J. Phytopathol. 115, 222–228.

- Heun M., Schäfer-Pregl R., Klawan D., Castagna R., Accerbi M., Borghi B., Salamini F. (1997) Site of Einkorn Wheat Domestication Identified by DNA Fingerprinting, Science 278, 1312–1314.
- Hidalgo A., Brandolini A., Pompei C., Piscozzi R. (2006) Carotenoids and tocols of einkorn wheat (*Triticum monococcum* ssp. *monococcum* L.), J. Cereal Sci. 44, 182–193.
- Hill J.P., Johnston M.R., Velasco V.R, Brown W.M. (1995) Responses of selected barley lines to barley stripe rust in Bolivia, Ecuador and Germany, Phytopathology 85, 1040.
- Hintum T.J.L., Knüpffer H. (1995) Duplication within and between germplasm collections. I. Identification duplication on the basis of passport data, Genet. Resour. Crop Ev. 42, 1127–1133.
- Hiura U. (1960) Studies on the disease-resistance in barley. IV. Genetics of resistance to powdery mildew, Ber. Ohara Inst. Landwirtsch. Biol. Okayama Univ. 11, 235–300.
- Hoffmann W., Nover I. (1959) Ausgangsmaterial f
  ür die Z
  üchtung mehltauresistenter Gersten, Z. Pflanzenz
  ücht. 42, 68–78.
- Holly L. (2000) Strategic questions of conserving agrobiodiversity in Hungary, Proceedings of the Hungarian Plant Breeding Conference, Hungarian Academy of Sciences, Budapest, p. 18.
- Honecker L. (1938) Über die physiologische Spezialisierung des Gerstenmehltaus als Grundlage für die Immunitätszüchtung, Züchter 10, 169–181.
- Horst W.J., Puschel A.K., Schmohl N. (1997) Induction of callose formation is a sensitive marker for genotypic aluminium sensivity in maize, Plant Soil 192, 23–30.
- Hsam S.L.K., Zeller F.J. (2002) Breeding for Powdery Mildew Resistance in common wheat (*Triticum aestivum* L.), in: Bélanger R.R., Bushnell W.R., Dik A.J., Carver T.L.W. (Eds.), The powdery mildews: A comprehensive treatise, The American Phytopathological Society, St. Paul, MN, USA, pp. 219–238.
- Hubert K.H., Buertsmayr H.B. (2006) Development of Methods for Bunt Resistance Breeding for Organic Farming, Czech J. Genet. Plant Breeding 42, 66–71.
- Igrejas G., Guedes-Pinto H., Carnide V., Clement J., Branlard G. (2002) Genetical, biochemical and technological parameters associated with biscuit quality. II. Prediction using storage proteins and quality characteristics in a soft wheat population, J. Cereal Sci. 36, 187–197.
- IPGRI (1985) Descriptors for Wheat (Revised), IPGRI, Rome, Italy.
- IPGRI (2003) A Guide to Effective Management of Germplasm Collections, in: Engels J.M.M., Visser L. (Eds.), IPGRI Handbooks for Genebanks 6, IPGRI, Rome, Italy.
- Jackson P., Robertson M., Cooper M., Hammer G. (1996) The role of physiological understanding in plant breeding; from a breeding perspective, Field Crop. Res. 49, 11–37.
- Jaradat A.A., Shahid M. (2006) Population and multilocus isozyme structures in a barley landrace, Plant Genet. Res. Charact. Util. 4, 108–116.
- Jaradat A.A., Ajlouni M.M., Duwayri M.A. (1996) Genetic resources of cereals and their wild relatives in Jordan: revisited, in: Jaradat A.A. (Ed.), Plant Genetic Resources of Jordan, IPGR, Regional Office for West Asia and North Africa, Aleppo, Syria, pp. 55–75.

- Jaradat A.A., Shahid M., Al Maskri A.Y. (2004) Genetic Diversity in the Batini Barley Landrace from Oman: I. Spike and Seed Quantitative and Qualitative Traits, Crop Sci. 44, 304–315.
- Jilal A., Grando S., Henry R.J., Lee L.S., Rice N., Hill H., Baum M., Ceccarelli S. (2008) Genetic diversity of ICARDA's worldwide barley landrace collection, Genet. Resour. Crop Ev. 2008, 1–10.
- Jin Y., Steffenson B.J., Bockelman H.E. (1995) Evaluation of cultivated and wild barley for resistance to pathotypes of *Puccinia hordei* with wide virulence, Genet. Resour. Crop Ev. 42, 1–6.
- Johnson V.A., Mattern P.J., Schmidt J.W. (1967) Nitrogen relations during spring growth in varieties of *Triticum aestivum* L. differing in grain protein content, Crop Sci. 7, 664–667.
- Jones H., Hinchsliffe K., Clarke S.M., Pearce B., Gibbon D., Lyon F., Harris F., Thomas J., Wolfe M.S. (2006) A participatory methodology for large scale field trials in the UK, in: Proceedings of the Joint Organic Congress, Odense, Denmark, May 30–31, 2006.
- Kandawa-Schulz M.A. (1996) Untersuchungen an Artbastarden von Wild- und Kulturgersten (*Hordeum bulbosum x H. vulgare*) zur Introgression züchterisch wichtiger Resistenzeigenschaften, Diss. Uni Rostock, Mathem.-Naturwiss. Fak.
- Kapulnik Y., Kushnir U. (1991) Growth dependency of wild, primitive and modern cultivated wheat lines on vesiculararbuscolar mycorrhizal fungi, Euphytica 56, 27–36.
- Kapulnik Y., Feldmann M., Okon Y., Henis Y. (1985) Contribution of nitrogen fixed Azospirillum to the nutrition of spring wheat in Israel, Soil Biol. Biochem. 17, 509–515.
- Kapulnik Y., Okon Y., Henis Y. (1987) Yield response of spring wheat cultivars (*Triticum aestivum* and *T. turgidum*) to inoculation with *Azospirillum brasiliense* under field conditions, Biol. Fert. Soils 4, 27–35.
- Kapulnik Y., Sarig S., Nur I., Okon Y. (1983) Effect of Azospirillum inoculation on yield of field-grown wheat, Can. J. Microbiol. 29, 895–899.
- Kashiwazaki S., Ogawa K., Usugi T., Tsuchizaki T. (1989) Characterization of several strains of barley yellow mosaic virus, Ann. Phytopathol. Soc. Japan 55, 16–25.
- Katsiotis A., Germeier C.U., Koenig J., Legget M., Bondo L., Frese L., Bladenopoulos K., Ottoson F., Mavromatis A., Veteläinen M., Menexes G., Drossou A. (2009) Screening a European Avena landrace collection using morphological and molecular markers for quality and resistance breeding, in: Proceeding of the EUCARPIA Cereals Section, Lleida, Spain, November 13–17, 2006.
- Kebebew F., Tsehaye Y., McNeilly T. (2001) Diversity of durum wheat (*Triticum durum* Desf.) at in situ conservation sites in North Shewa and Bale, Ethiopia, J. Agric. Sci. Camb. 136, 383–392.
- Khatskevitch L.K., Benken A.A. (1990) Root and stem rots of cereals, Zashchita rastenii, Moskva 9, 14–15.
- Khokhlova A.P., Vershinina V.A., Lukyanova M.V., Terentyeva I.A. (1989) VIR Catalogue: Characteristics of susceptibility to loose smut, leaf rust and powdery mildew, VIR, St. Petersburg, 30 p.
- Kirdoglo E.K. (1990) Breeding barley for resistance to smut and leaf/stem diseases, Vestnik Selskokhozyaistvennoi Nauki Moskva 10, 98–104.

- Kmecl A., Mauch F., Winzeler H., Dudler R. (1995) Quantitative field resistance of wheat to powdery mildew and defence reactions at the seedling stage-identification of a potential marker, Physiol. Mol. Plant P. 47, 185–199.
- Kochian L.V., Pineros M.A., Hoekenga O.A. (2005) The physiology, genetics and molecular biology of plant aluminium resistance and toxicity, Plant Soil 274, 175–195.
- Kolodinska Brantestam A.K., Von Bothmer R., Dayteg C., Rashal I., Tuvesson S. Weibull J. (2004) Inter simple sequence repeat analysis of genetic diversity and relationships in cultivated barley of Nordic and Baltic origin, Hereditas 141, 186–192.
- Kolodinska Brantestam A.K., Von Bothmer R., Rashal I., Weibull J. (2003) Changes in the genetic diversity of barley of Nordic and Baltic origin, studied by isozyme electrophoresis, Plant Genet. Res. Charact. Util. 1, 143–149.
- Kotali E. et al. (2008) 12th meeting of the Hellenic Scientific Society for Plant Breeding and Genetics, Naoussa 8–10 October 2008, p. 121.
- Kovács G. (2006a) Selection of field crop varieties suitable for organic farming, Biokontroll Hungária Kht. Budapest, p. 88.
- Kovács G. (2006b) The possible use of founder effect to produce locally adapted cereal varieties, in: Oestergard H., Fontaine L. (Eds.), Cereal Crop Diversity: Implications for production and products ITAB, Paris, France, pp. 68–70.
- Kovács G. (2008) Bread wheat composite crosses as a source modern landraces, NKTH, Budapest, Research Report, p. 95.
- Kovács G., Kótai C., Kanyó Z., Láng L., Nagyne K.R., Polgáár Z., Roszík P., Zsigrai G. (2006) Variety recommendation list for organic farmers, in: Kovács G. (Ed.), Selection of field crop varieties suitable for organic farming, Biokontroll Hungária Kht. Budapest, pp. 39–82 (in Hungarian).
- Kuhlmann H., Barraclough P.B., Weir A.H. (1989) Utilization of mineral nitrogen in the subsoil by winter wheat, Z. Pflanzenernähr. Bodenk. 152, 291–295.
- Kurilich A.C., Juvik J.A. (1999) Quantification of carotenoid and tocopherol antioxidants in *Zea mays*, J. Agr. Food Chem. 47, 1948–1955.
- Láng L. (2006) Improving the yield, adaptability and protein content of barley, wheat and spelt, in: Bedõ Z., Kovács G. (Eds.), Organic breeding and cultivation of cereals, Agroinform Publishing Ltd. Budapest, pp. 88–93.
- Laperche A., Devienne-Barret F., Maury O., Legouis J., Ney B. (2006) A simplified conceptual model of carbon/nitrogen functioning for QTL analysis of winter wheat adaptation to nitrogen deficiency, Theor. Appl. Genet. 113, 1131–1146.
- Larsson H. (2006) Old cultural cereal varieties are broadening the genetic base for organic farming and will increase the quality for consumers, Proceedings of the ECO-PB Workshop: "Participatory Plant Breeding: Relevance for Organic Agriculture?", La Besse, France 11–13 June 2006.
- Lehmann L., Bothmer R. von, Jorna M.L., Slootmaker L.A.J. (1988) *Hordeum spontaneum* and landraces as a gene resource for barley breeding. Cereal breeding related to integrated cereal production, Proceedings of the conference of the Cereal Section of EUCARPIA, Wageningen, Netherlands, 24–26 February 1988, pp. 190–194.
- Leur Van J.A.G. Pathology, Cereal Improvement Program Ann. Rep., 1988, pp. 122–130.

- Little R. (1988) Plant soil interactions at low pH, in: Problem Solving – The Genetic Approach, Commun. Soil Sci. Plan. Anal. 19, 1239–1257.
- Lukyanova M.V. (1990) New sources and donors for barley breeding, Byulleten Vsesoyuznogo Nauchno Issledovatel'skogo Instituta Rastenievodstva imeni N.I. Vavilova 201, 26–30.
- Lukyanova M.V., Terentyeva I.A. (1997) Catalogue of genetic collection of barley with identified genes of resistance to powdery mildew, 48 VIR, St. Petersburg, Russia.
- Luthra J.K., Verma R.S., Prabhu K.V. (1992) Development of isogenic lines in barley with reference to *Puccinia striiformis* West, in: Munck L. (Ed.), Barley Genet. VI, 622–625, Munksgaard Int. Publ. Ltd., Copenhagen, Denmark.
- Mamluk O.F., Nachit M.M. (1994) Sources of Resistance to Common Bunt (*Tilletia foetida* and *T. caries*) in Durum Wheat, J. Phytopathol. 142, 122–130.
- Manjunatha T., Bisht I.S., Bhat K.V., Singh B.P. (2007) Genetic diversity in barley (*Hordeum vulgare* L. ssp. *vulgare*) landraces from Uttaranchal Himalaya of India, Genet. Resour. Crop Ev. 54, 55–65.
- Mannerstedt-Fogelfors B. (2001) Antioxidants and lipids in oat cultivars as affected by environmental factors, Doctoral Thesis, Acta Universitatis Agriculturae Sueciae, Agraria 271, ISSN 1401–6249, Uppsala.
- Mano Y., Takeda K. (1995) Varietal variation and effects of some major genes on salt tolerance in barley seedlings, Okayama University, Bull. Res. Inst. Biores. 3, 71–81.
- Mano Y., Nakazumi H., Takeda K. (1996) Varietal variation in and effect of some major genes on salt tolerance at the germination stage in barley, Breeding Sci. 46, 227–233.
- Manske G.G.B. (1989) Genetical Analysis of VA mycorrhiza with spring wheat, Agr. Ecosyst. Environ. 29, 273–280.
- Manske G.G.B. (1990) Genetical analysis of the efficiency of VA mycorrhiza with spring wheat, in: El Bassam et al. (Eds.), Genetic aspects of plant mineral nutrition, 397–405, Kluwer Academic Publishers.
- Manske G.G.B., Lüttger A.B., Behl R.K., Vlek P.L.G. (1995) Nutrient Efficienca Based on VA Mycorrhizae (VAM) and total root length of Wheat Cultivars Grown in India, Angew. Bot. 69, 108–110.
- Mantzavinou A., Bebeli P.J., Kaltsikes P.J. (2005) Estimating genetic diversity in Greek durum wheat landraces with RAPD markers, Aust. J. Agr. Res. 56, 1355–1364.
- Martínez F., Rubiales D. (2002) Resistance to leaf rust in durum wheat cultivar Creso, Cereal Rusts and Powdery Mildews Bull. 30, 2002, http://www.crpmb.org/2002/1130martinez.
- Martínez F.B., Niks R.E., Rubiales D. (2001a) Partial resistance to leaf rust in a collection of ancient Spanish barleys, Hereditas 135, 199–203.
- Martínez F., Niks R.E., Moral A., Urbano J.M., Rubiales D. (2001b) Search for partial resistance to leaf rust in a collection of ancient Spanish wheats, Hereditas 135, 193–197.
- Martínez F., Niks R.E., Singh R.P., Rubiales D. (2001c) Characterization of *Lr46*, a gene conferring partial resistance to wheat leaf rust, Hereditas 135, 111–114.
- Martos V., Royo C., Rharrabti Y., Garcia del Moral L.F. (2005) Using AFLPs to determine phylogenetic relationships and genetic erosion in durum wheat cultivars released in Italy and Spain throughout the 20th century, Field Crop. Res. 91, 107–116.

- Masum Akond A.S.M.G., Watanabe N. (2005) Genetic variation among Portuguese landraces of "Arrancada" wheat and *Triticum petropavlovskyi* by AFLP-based assessment, Genet. Resour. Crop Ev. 52, 619–628.
- Matsuoka Y., Vigouroux Y., Goodman M.M., Sanchez G.J., Buckler E., Doebley J. (2002) A single domestication for maize shown by multilocus microsatellite genotyping, Proc. Natl Acad. Sci. (USA) 99, 6080–6084.
- Maxted N. (2003) Conserving the genetic resources of crop wild relatives in European Protected Areas, Biol. Conserv. 113, 411–417.
- McClintock B. (1984) The significance of the responses of the genome to challenge, Science 226, 792–801.
- McIntosh R.A., Hart G.E., Devos K.M., Gale M.D., Rogers W.J. (1998) Catalogue of gene symbols for wheat, Vol. 5, in: Proc. 9th Int. Wheat Genet. Symp., Saskatchewan, Canada, pp. 134–138.
- Medini M., Hamza S., Rebai A., Baum M. (2005) Analysis of genetic diversity in Tunisian durum wheat cultivars and related wild species by SSR and AFLP markers, Genet. Resour. Crop Ev. 52, 21–31.
- Miflin B. (2000) Crop Improvement in the 21st century, J. Exp. Bot. 51, 1–8.
- Monasterio J.I., Graham R.D. (2000) Breeding for trace minerals in wheat, Food Nutr. Bull. 21, 393–396.
- Moore-Colyer R.J. (1995) Oats and oat production in history and pre-history, in: Welch R.W. (Ed.), The oat crop, Chapman & Hall, pp. 1–33.
- Moragues M., Garcia del Moral L.F., Moralejo M., Royo C. (2006a) Yield formation strategies of durum wheat landraces with distinct pattern of dispersal within the Mfeiterranean basin. I. Yield components, Field Crop. Res. 95, 194–205.
- Moragues M., Garcia del Moral L.F., Moralejo M., Royo C. (2006b) Yield formation strategies of durum wheat landraces with distinct pattern of dispersal within the Mediterranean basin. II. Biomass production and allocation, Field Crop. Res. 95, 182–193.
- Moreira P.M. (2006) Participatory maize breeding in Portugal. A case study, Acta Agron. Hungarica 54, 431–439.
- Morrell P.L., Toleno D.M., Lundy K.E., Clegg M.T. (2005) Low levels of linkage disequilibrium in wild barley (*Hordeum vulgare ssp. spontaneum*) despite high rates of self-fertilization, Proc. Natl Acad. Sci. (USA) 102, 2442–2447.
- Morris M.L., Bellon M.R. (2004) Participatory plant breeding research: Opportunities and challenges for the international crop improvement system, Euphytica 136, 21–35.
- Moseman J.G. (1955) Sources of resistance to powdery mildew of barley, Plant Dis. Rep. 39, 967–972.
- Moseman J.G., Craddock J.C. (1976) Genetic basis for barley germ-plasm conservation, in: Gaul (Ed.), Barley Genetics III, Proc. Third Int. Barley Genet. Sympos., Verlag Karl Thieming, Munich, pp. 51–57.
- Moseman J.G., Smith R.T. (1976) Breeding for multiple pest resistance, Barley Genet. III, Proc. 3rd Int. Barley Gen. Sympos., Garching, in: Gaul H. (Ed.), Verlag Karl Thieming, Munich, pp. 421–425.
- Motzo R., Giunta F. (2007) The effect of breeding on the phenology of Italian durum wheats: From landraces to modern cultivars, Eur. J. Agron. 26, 462–470.
- Munns R. (2005) Response of crops to salinity, in: International salinity forum - managing saline soils and water: science, technology and social issues, Riverside Convention Center,

Riverside, California, USA, 25–28 April 2005, pp. 339–342, USDA-ARS Salinity Laboratory, Riverside, USA.

- Murphy K., Jones S. (2006) Breeding wheat for organic conditions, Nutritional value, http://www.organic-center.org/ reportfiles/Jones%20Tilth%2006.pdf.
- Murphy K.M., Reeves P.G., Jones S.S. (2008) Relationship between yield and mineral nutrient concentrations in historical and modern spring wheat cultivars, Euphytica 163, 381–390.
- Nachit M.M., Ketata H., Yau S.K. (1988) Breeding durum wheat for stress environments of the Mediterranean region, in: Wittner. G. (Ed.), The future of cereals for human feeding and development of biotechnological research, Proc. Third Int. Symp. on durum wheat, Foggia, Italy, 5–7 May 1988.
- Nawrot M., Szarejko I., Maluszynski M. (2001) Barley mutants with increased tolerance to aluminium toxicity, Euphytica 120, 345–356.
- Negassa M. (1985a) Geographic distribution and genotypic diversity of resistance to powdery mildew of barley in Ethiopia, Hereditas 102, 113–121.
- Negassa M. (1985b) Genetics of resistance to powdery mildew in some Ethiopian barleys, Hereditas 102, 123–138.
- Nelson K.E., Burgess L.W. (1994) Reaction of Australian cultivars of oats and barley to infection by *Fusarium graminearum* Group 1, Aust. J. Exp. Agr. 34, 655–658.
- Nettevich E.D., Vlasenko N.M. (1985) Results of evaluating the barley collection in the central nonchernozem zone, Sbornik Nauchn. Trud. Prikl. Bot. Gen. Selektsii 95, 28–31.
- Nevo E. (1992) Origin, evolution, population genetics and resources of wild barley, Hordeum spontaneum, in the Fertile Crescent, in: Shewry P.R. (Ed.), Barley: Genetics, Biochemistry, Molecular Biology and Biotechnology, CAB International, Wallingford, UK, pp. 19–43.
- Newton A.C. (1990) Detection of components of partial resistance to mildew (*Erysiphe graminis f. sp. hordei*) incorporated into advanced breeding lines of barley using measurement of fungal cell wall sterol, Plant Pathol. 39, 598–602.
- Newton A.C., Thomas W.T.B. (1994) Detection of tolerance of barley cultivars to infection by powdery mildew (*Erysiphe* graminis f.sp. hordei), Euphytica 75, 179–187.
- Newton A.C., Begg G., Swanston J.S. (2009) Deployment of diversity for enhanced crop function, Ann. Appl. Biol. 154, 309–322.
- Newton A.C., Guy D.C., Gaunt R.E., Thomas W.T.B. (2000) The effect of powdery mildew inoculum pressure and fertilizer levels on disease tolerance in spring barley, J. Plant Dis. Prot. 107, 67–73.
- Newton A.C., Thomas W.T.B., Guy D.C., Gaunt R.E. (1998) The interaction of fertiliser treatment with tolerance to powdery mildew in spring barley, Field Crop. Res. 55, 45–56.
- Niks R.E., Rubiales D. (2002) Detection of potentially durable resistance mechanisms in plants to specialised fungal pathogens, Euphytica 124, 201–216.
- Nordborg M., Borevitz J.O., Bergelson J., Berry C.C., Chory J., Hagenblad J., Kreitman M., Maloof J.N., Noyes T., Oefner P.J., Stahl E.A., Weigel D. (2002) The extent of linkage disequilibrium in *Arabidopsis thaliana*, Nat. Genet. 30, 190–193.
- Nover I., Lehmann C.O. (1966) Resistenzigenschaften im Gersten- und Weizensortiment Gatersleben 6. Prüfung

von Gersten auf ihr Verhalten gegen Gelbrost (*Puccinia striiformis* West. syn. P. glumarum (Schm.) Erikss. et Henn.), Die Kulturpflanze 14, 257–262.

- Nover I., Lehmann C.O. (1970) Resistance characters in the barley and wheat collection gatersleben [Resistenzeigenschaften im Gersten - und Weizensortiment Gatersleben - 13. Prüfung von Wintergersten-Neuzugängen Auf Ihr Verhalten Gegen Gelbrost, *Puccinia Striiformis* West], Die Kulturpflanze 18, 107–108.
- Nover I., Lehmann C.O. (1974) Resistance in the barley and wheat collection Gatersleben. 18. Screening spring barleys for their reaction to leaf rust (*Puccinia hordei* Otth.) [Resistenzeigenschaften im Gersten- und Weizensortiment Gatersleben 18. Prüfung von Sommergersten auf ihr Verhalten gegen Zwergrost (*Puccinia hordei* Otth)], Die Kulturpflanze 22, 25–43.
- Nover I., Lehmann C.O. (1975) Resistance in the barley and wheat collection Gatersleben. 19. Testing spring barleys for their reaction to race 24 of stripe rust (*Puccinia striiformis* West.) [Resistenzeigenschaften im Gersten - und Weizensortiment Gatersleben - 19. Prüfung von Sommergersten auf ihr Verhalten gegen Gelbrost, *Puccinia striiformis* West., Rasse 24], Die Kulturpflanze 23, 75–81.
- Nover I., Mansfeld R. (1955) Resistenzeigenschaften im Gersten und Weizensortiment Gatersleben, Die Kulturpflanze 3, 105–113.
- Nover I., Mansfeld R. (1956) Resistenzeigenschaften im Gersten und Weizensortiment Gatersleben II, Die Kulturpflanze 4, 341–349.
- Nover I., Lehmann C.O., Seidenfaden A. (1976) Resistance in the Gatersleben barley and wheat collection 20. Testing of barley for reaction to loose smut *Ustilago nuda*, Kulturpflanze 24, 237–248.
- Oak M.D., Tamhankar S.A., Rao V.S., Bhosale S.B. (2004) Relationship of HMW, LMW glutenin subunits and  $\gamma$ -gliadins with gluten strength in Indian durum wheats, J. Plant Biochem. Biot. 13, 51–55.
- Okunowski I. (1990) Untersuchungen von Gersten aus dem Gaterslebener Weltsortiment auf Resistenz gegen Gelbrost *Puccinia striiformis* West. unter besonderer Berücksichtigung der Feldresistenz, Diss. Akad. Landwirtschaftswissenschaften, Berlin.
- Onishkova M.G. (1987) Immunity characteristics in Asian barley in the northern Kazakhstan, Selektsionno-Geneticheskie Issledovanijya pri Vyvedenii Novykh Sortov Polevykh Kul'tur Severnogo-Kazkhstana, 89–92.
- Oury F.-X., Leenhardt F., Rémésy C., Chanliaud E., Duperrier B., Balfourier F., Charmet G. (2006) Genetic variability and stability of grain magnesium, zinc and iron concentration in bread wheat, Eur. J. Agron. 25, 177–185.
- Pagnotta M.A., Impiglia A., Tanzarella O.A., Nachit M.M., Porceddu E. (2004) Genetic variation of the durum wheat landrace Haurani from different agro-ecological regions, Genet. Resour. Crop Ev. 51, 863–869.
- Panfili G., Fratianni A., Irano M. (2004) Improved Normal-Phase High-Performance Liquid Chromatography Procedure for the Determination of Carotenoids in Cereals, J. Agr. Food Chem. 52, 6373–6377.
- Panga X.P., Lethaya J. (2000) Challenge of Timing Nitrogen Availability to Crop Nitrogen Requirements, Soil Sci. Soc. Am. J. 64, 247–253.

- Papa R., Attene G., Barcaccia G., Ohgata A., Konishi T. (1998) Genetic diversity in landrace populations of *Hordeum vulgare* L. from Sardinia, Italy, as revealed by RAPDs, isozymes and morphophenological traits, Plant Breeding 117, 523–530.
- Papakosta D.K., Garianas A.A. (1991) Nitrogen and dry matter accumulation, remobilisation, and losses for Mediterranean wheat during grain filling, Agron. J. 83, 864–870.
- Parry M.A.J., Shewry P.R. (2003) Genotype-phenotype: narrowing the gap, Ann. Appl. Biol. 142, 115.
- Passiuora J.B. (1996) Drought and drought tolerance, Plant Growth Regul. 20, 79–83.
- Pecetti L., Boggini G., Gorham J. (1994) Performance of durum wheat landraces in a Mediterranean environment (eastern Sicily), Euphytica 80, 191–199.
- Pecetti L., Damania A.B., Jana J. (1992) Practical problems in large-scale germplasm evaluation. A case study in durum wheat, FAO/IBPGR PI, Genet. Res. Newsl. 88/89, 5–10.
- Pecetti L., Doust M.A., Calcagno L., Paciti C.N., Boggini G. (2002) Variation of morphological and agronomical traits and protein composition in durum wheat germplasm from Eastern Europe, Genet. Resour. Crop Ev. 48, 601–620.
- Pecetti L., Doust M.A., Calcagno L., Raciti C.N., Boggini G. (2001) Variation of morphological and agronomical traits, and protein composition in durum wheat germplasm from Eastern Europe, Genet. Resour. Crop Ev. 48, 609–620.
- Pedrechi R., Cisneros-Zevallos L. (2006) Antimutagenic and antioxidant properties of phenolic fractions from Andean purple corn (Zea mays L.), J. Agr. Food Chem. 54, 4557–4567.
- Perkowski J., Kiecana I., Chelkowski J. (1995) Susceptibility of barley cultivars and lines to *Fusarium* infection and mycotoxin accumulation in kernels. J. Phytopathol. 143, 547–551.
- Perkowski J., Kiecana I., Schumacher U., Muller H.M., Chelkowsi J., Golinski P. (1997) Head infection and accumulation of *Fusarium* toxin in kernels of 12 barley genotypes inoculated with *Fusarium graminearum* isolates of two chemotypes, Eur. J. Plant Pathol. 103, 85–90.
- Peterson D.M. (2004) Oat a multifunctional grain, Proceedings of the 7th International Oat Conference, Helsinki, Finland, in: Peltonen-Sainio P., Topi-Hulmi M. (Eds.), Jokioinen: MTT Agrifood Research Finland, Agrifood Res. Rep. 51, 21–25.
- Peusha H., Lebedeva T., Prilinn O., Enno T. (2002) Genetic analysis of durable powdery mildew resistance in a common wheat line, Hereditas 136, 201–206.
- Piffanelli P., Ramsay L., Waugh R., Benabdelmouna A., D'Hont A., Hollricher K., Jørgensen Jø.H., Schulze-Lefert P., Panstruga R. (2004) A barley cultivation-associated polymorphism conveys resistance to powdery mildew, Nature 430, 887–891.
- Pinheiro de Carvalho M.Â.A., Slaski J.J., Abreu I., Ganança F.T., dos Santos T.M.M., Freitas L., Clemente Vieira M.R., Nunes A., Domingues A., Taylor G.J. (2004) Factors contributing to the development of aluminium resistance in the Madeiran maize germplasm, J. Plant Nutr. Soil Sc. 167, 93–98.
- Pinheiro de Carvalho M.Â.A., Slaski J.J., dos Santos T.M.M., Ganança F.T., Abreu I., Taylor G.J., Clemente Vieira M.R., Popova T.N., Franco E. (2003) Identification of aluminium resistant genotypes among madeiran regional wheats, Commun. Soil Sci. Plan. 34, 2967–2979.

- Pinheiro de Carvalho M.A.A., Ganança I.F.T., Abreu I., Sousa N.F., dos Santos T.M.M., Vieira Clemente R.M., Motto M. (2008) Evaluation of the maize (Zea mays L.) diversity on the Archipelago of Madeira, Genet. Resour. Crop Ev. 55, 221–233.
- Pommer G. (1990) Accumulation and translocation of nitrogen in cultivars of wheat with different demands for nutrition, in: El Bassam et al.(Eds.), genetic aspects of plant nutrition, Kluwer Academic Publishers, 1990.
- Proeseler G., Hartleb H., Kopahnke D., Lehmann Chr. O. (1989) Resistenzeigenschaften im Gersten - und Weizensortiment Gatersleben - 29. Pr
  üfung von Gersten auf ihr Verhalten gegen
  über dem Milden Gerstenmosaik-Virus (barley mild mosaic virus, BaMMV), Gerstengelbmosaik-Virus (barley yellow mosaic virus, BaYMV), Drechslera teres (Sacc.) Shoem. und Puccinia hordei Otth., Die Kulturpflanze 37, 145–154.
- Queen R.A., Gribbon B.M., James C., Jack P., Favell A.J. (2004) Retrotransposon-based molecular markers for linkage and genetic diversity analysis in wheat, Mol. Genet. Genomics 271, 91–97.
- Raciti C.N., Doust M.A., Lombardo G.M., Boggini G., Pecetti L. (2003) Characterization of durum wheat mediterranean germplasm for high and low molecular weight glutenin subunits in relation with quality, Eur. J. Agron. 19, 373–382.
- Rakszegi M., Láng L., Bedõ Z. (2006) Selection of wheat genotypes adaptable to organic farming conditions using classical and molecular breeding methods, in: Bedõ Z., Kovács G. (Eds.), Organic breeding and cultivation of cereals, Agroinform Publishing Ltd. Budapest, pp. 83–88.
- Rasmusson D.C., Philips R.L. (1997) Plant breeding progress and genetic diversity from de novo variation and elevated epistasis, Crop Sci. 37, 303–310.
- Rebourg C., Chastanet M., Gouesnard B., Welcker C., Dubreuil P., Charcosset A. (2003) Maize introduction into Europe: the history reviewed in the light of molecular data, Theor. Appl. Genet. 106, 895–903.
- Rebourg C., Gouesnard B., Charcosset A. (2001) Large scale molecular analysis of traditional European maize populations. Relationships with morphological variation, Heredity 86, 574–587.
- Reif J.C., Hamrit S., Heckenberger M., Schipprack W., Maurer H.P., Bohn M., Melchinger A.E. (2005) Genetic structure and diversity of European flint maize populations determined with SSR analyses of individuals and bulks, Theor. Appl. Genet. 111, 906–913.
- Reinhold M., Sharp E.L. (1986) Resistance sources in barley to Puccinia hordei, Cereal Rusts Bull. 14, 75–83.
- Richards R.A. (1982) Breeding and selecting for drought resistance in wheat, in: Drought Resistance in Crops with Emphasis on Rice, IRRI, Los Banos, Philippines, pp. 303–316.
- Rigina S.I. (1966) Izuchenie ustoychivosti yachmenya k infekcionnym zabolevaniyam [Study of resistance of barley to infectious diseases], Can. Agr. Sci. Diss., Leningrad, 196 p.
- Rochev M.V., Levitin M.M. (1986) Resistance of barley varieties to *Bipolaris sorokiniana* (Sacc.) Shoemaker in the central Urals, Sibirski Vestnik Selskokhozyaistvennoi Nauk 6, 18–21.
- Roelfs A.P. (1988) Resistance to leaf and stem rust of wheat, in: Simmonds N.W., Rajaram S. (Eds.), Breeding strategies for resistance to the rusts of wheat, México CIMMYT.

- Rubiales D., Niks R.E. (2000) Combination of mechanisms of resistance to rust fungi as a strategy to increase durability, Opt. Méditerranéennes 40, 333–339.
- Ruiz M., Aguiriano E. (2004) Analysis of duplication in the Spanish durum wheat collection maintained in the CRF-INIA on the basis of agro-morphological traits and gliadin proteins, Genet. Resour. Crop Ev. 51, 231–235.
- Ruiz M., Martín I. (2000) Spanish landraces collection of durum wheat maintained at the CRF-INIA, Opt. Mediterraneennes Serie A 40, 601–606.
- Ruiz De Galarreta J.I., Alvarez A. (2001) Morphological classification of maize landraces from northern Spain, Genet. Resour. Crop Ev. 48, 391–400.
- Russell J.R., Booth A., Fuller J.D., Baum M., Ceccarelli S., Grando S., Powell W. (2003) Patterns of polymorphism detected in the chloroplast and nuclear genomes of barley landraces sampled from Syria and Jordan, Theor. Appl. Genet. 107, 413–421.
- Sabri N., Dominy P.J., Clarke D.D. (1997) The relative tolerances of wild and cultivated oats to infection by *Erysiphe* graminis f.sp. avenae: II. The effects of infection on photosynthesis and respiration, Physiol. Mol. Plant P. 50, 321–335.
- Sackville Hamilton N.R., Chorlton K.H. (1997) Regeneration of Accessions in Seed Collections: a Decision Guide, Handbook for Genebanks No. 5, International Plant Genetic Resources Institute, Rome.
- SANCO (2006) Commission directive of providing for certain derogations for acceptance of agricultural landraces and cultivars which are naturally adapted to the local and regional conditions and threatened by genetic erosion and for marketing of seed and seed potatoes of those landraces and cultivars, SANCO/3322/2006 Rev. 14.
- Sato K., Takeda K. (1994) Sources of resistance to net blotch in barley germplasm, Bull. Res. Inst. Bioresour., Okayama Univ. 2, 91–102.
- Schaller C.W., Wiebe G.A. (1952) Sources of resistance to net blotch of barley, Agron. J. 44, 334–336.
- Schaller C.W., Qualset C.O., Rutger J.N. (1964) Inheritance and linkage of the Yd2 gene conditioning resistance to barley yellow dwarf virus disease in barley, Crop Sci. 4, 544–548.
- Semeane Y. (1995) Importance and control of barley leaf blights in Ethiopia, Rachis 14, 83–89.
- Shchelko L.G. (1969) Study of initial material of barley for resistance to loose smut, Sp. tr. V Vsesoyuzn. soveshchaniya po immunitetu (Kiev) 5, 7–12.
- Shiva V. (2007) Manifestos on the future of food and seed, Cambridge, Massachusetts.
- Shtaya M.J.Y., Sillero J.C., Rubiales D. (2006a) Screening for resistance to leaf rust (*Puccinia hordei*) in a collection of Spanish barleys, Breeding Sci. 56, 173–177.
- Shtaya M.J.Y., Sillero J.C., Rubiales D. (2006b) Search for partial resistance against *Puccinia hordei* in barley landraces from Fertile Crescent, Plant Breeding 125, 343–346.
- Shtaya M.J.Y., Sillero J.C., Rubiales D. (2006c) Identification of resistance against a new pathotype of *Puccinia hordei* with virulence for the resistance gene *Rph7*, Eur. J. Plant Pathol. 115, 309–321.
- Siddique K.H.M., Belfort R.K., Tennant D. (1990) Root:shoot ratios of old and modern, tall and semidwarf wheats in mediterranean environment, Plant Soil 121, 89–98.

- Simmonds N.W. (1979) Principles of Crop Improvement, Longman, New York, p. 408.
- Skou J.P., Haahr V. (1985) The barleys in Nordic Gene Bank screened for resistance against barley leaf stripe (*Drechslera* graminea), Nordisk Jordbrugsforskning 67, 262–263.
- Skou J.P., Nielsen B.J., Haahr V. (1992) The effectivity of Vada resistance against leaf stripe in barley varieties, Nordisk Jordbrugsforskning 74, 34.
- Skou J.P., Nielsen B.J., Haahr V. (1994) Evaluation and importance of genetic resistance to leaf stripe in western European barleys, Acta Agr. Scand. B, Soil Plant Sci. 44, 98–106.
- Slaski J.J. (1992) Physiological and genetical aspects of the tolerance of cereals to soil acidity and to toxic effects of aluminium ions, Bull. IHAR 183, 37–45.
- Smirnova Z. G., Trofimovskaya A.Ya. (1985) Sources of resistance of barley to net blotch, Sb. Nauchn. Tr. Prikl. Bot., Benet. i Sel. 95, 52–56.
- Stodart B.J., Raman H., Coombes N., Mackay M. (2007) Evaluating landraces of bread wheat *Triticum aestivum* L. for tolerance to aluminium under low pH conditions, Genet. Resour. Crop Ev. 54, 759–766.
- Stubbs R.W. (1985) The Cereal rusts, Vol. 2, Disease, Epidemiology and Control, in: Roelfs A., Bushnell W.R. (Eds.), Orlando, Academic Press, pp. 61–101.
- Su L., Maric A., Kostic B., Mikic K. (1989) Some properties of *Helminthosporium gramineum* and barley resistance to the pathogen, Zastita Bilja 40, 151–164.
- Surin N.A. (1989) Source material and problems of breeding spring barley in the eastern Siberia, Sbornik Nauchn. Trud. Prikl. Bot. Gen. Selektsii 129, 37–41.
- Swaminathan M.S. (2002) The Past, Present and Future Contributions of Farmers of the Conservation and Development of Genetic Diversity, IPGRI 2002, Managing Plant Genetic Diversity, in: Engels J.M.M., Ramanatha Rao V., Brown A.H.D., Jackson M.T., Cabi Publishing, Wallingford, UK, pp. 23–31.
- Takeda K. (1992) Current topics on the scab disease resistance in barely and wheat, Proc. Japan Assoc. Mycotoxicol. 36, 13–17.
- Takeda K., Heta H. (1989) Establishing the testing method and a search for the resistant varieties to *Fusarium* head blight in barley, Japan J. Breed. 39, 203–216.
- Tang C., Asseng S., Diatloff E., Rengel Z. (2003) Modelling yield losses of aluminium-resistant and aluminium-sensitive wheat due to subsurface soil acidity: effects of rainfall, liming and nitrogen application, Plant Soil 254, 349–360.
- Teklu Y., Hammer K., Huang X.Q., Röder M.S. (2005) Analysis of microsatellite diversity in Ethiopian tetraploid wheat landraces, Genet. Resour. Crop Ev. 100, 1–12.
- Tenaillon M.I., Sawkins M.C., Long A.D., Gaut R.L., Doebley J.F., Gaut B.S. (2001) Patterns of DNA sequence polymorphism along chromosome 1 of maize (*Zea mays* ssp. *mays* L.), Proc. Natl Acad. Sci. (USA) 98, 9161–9166.
- Teshome A., Brown A.H.D., Hodgkin T. (2001) Diversity in landraces of cereal and legume crops, in: Janick J. (Ed.), Plant Breeding Reviews, John Wiley & Sons, Inc., pp. 221–261.
- Thomas W.T.B. (2003) Prospects for molecular breeding of barley, Ann. Appl. Biol. 142, 1–12.
- Tilman D. (1996) The greening of the green evolution, Nature 396, 211–212.

- Tokatlidis I.S., Koutsika-Sotiriou M., Fasoulas A.C. (2001) The development of density-independent hybrids in maize, Maydica 46, 21–25.
- Tollenaar M., Wu J. (1999) Yield improvement in temperate maize is attributable to greater stress tolerance, Crop Sci. 39, 1597–1604.
- Uauy C., Distelfeld A., Fahima T., Blechl A., Dubcovsky J. (2006) A NAC Gene Regulating Senescence Improves Grain Protein, Zinc, and Iron Content in Wheat, Science 314, 1298–1301.
- Upadhyay M.K., Prakash S. (1977) Identification of diverse genes conferring resistance to indian races of stripe rust of barley, Indian J. Genet. Plant Breeding 37, 68–72.
- Valamoti S.M. (2002) Food remains from Bronze Age Archondiko and Mesimeriani Toumba in northern Greece? Veg. Hist. Archaeobot. 11, 17–22.
- Van Beem J. (1997) Variation in nitrogen use efficiency and root system size in temperate maize genotypes, in: Edmeades G.O., Banziger M., Mickelson H.R., Pena-Valdivia C.B. (Eds.), Developing drought- and low-N-tolerant maize, Cimmyt, Mexico.
- Van Dijk K.V., Parlevliet J.E., Kema G.H.J., Zeven A.C., Stubbs R.W. (1988) Characterization of the durable resistance to yellow rust in old winter wheat cultivars in the Netherlands, Euphytica 38, 149–158.
- Van Ginkel M., Rajaram S. (1992) Breeding for durable resistance to diseases in wheat: An international perspective, Durability of Disease Resistance, in: Jacobs Th., Parlevliet J.E. (Eds.), Kluwer Academic Publ., Dordrecht, pp. 259–272.
- Van Leur J.A.G. (1989) Barley pathology, Cereal Improvement Program Annual Report, 1988, pp. 122–130.
- Van Leur J.A.G., Ceccarelli S., Grando S. (1989) Diversity for disease resistance in barley landraces from Syria and Jordan, Plant Breeding 103, 324–335.
- Van Treuren R., Tchoudinova I., van Soest L.J.M., van Hintum T.J.L. (2006) Marker assisted acquisition and core collection formation: a case study in barley using AFLPs and pedigree data, Genet. Resour. Crop Ev. 53, 43–52.
- Vaz Patto M.C., Moreira P.M., Almeida N., Satovic Z., Pêgo S. (2008) Genetic diversity evolution through participatory maize breeding in Portugal, Euphytica 161, 283–291.
- Vaz Patto M.C., Satovic Z., Pêgo S., Fevereiro P. (2004) Assessing the genetic diversity of Portuguese maize germplasm using microsatellite markers, Euphytica 137, 63–72.
- Vechet L., Vojácková M. (2005) Use of Detached Seedling Leaf Test to Evaluate Wheat Resistance to Septoria Tritici Blotch, Czech J. Genet. Plant Breeding 41, 112–116.
- Velibekova E.L. (1981) Susceptibility of barley to root rots, Selektsiya i Semenovodstvo, USSR 10, 20–21.
- Wahl I., Anikster Y., Manisterski J. (1988) Evolution of hostparasite relations in the *Puccinia hordei - Hordeum* ssp. system at the center of origin, Proc. 7th Cereal Rusts Conference, Vienna, pp. 122–123.
- Waines J.G., Ehdaie B. (2007) Domestication and Crop Physiology: Roots of Green Revolution Wheat, Ann. Bot. 100, 991–998.
- Walther U., Lehmann C.O. (1980) Resistenzeigenschaften im Gersten - und Weizensortiment Gatersleben - 24. Prüfung von Sommer und Wintergersten auf ihr Verhalten geganuber Zwergrost (*Puccinia hordei* Otth), Die Kulterpfanze 28, 227–238.

- Weltzien E. (1988) Evaluation of barley (*Hordeum vulgare* L.) landrace populations originating from different growing regions in the Near East, Plant Breeding 101, 95–106.
- Wesenberg D.M., Briggle L.W., Smith D.H. (1992) Germplasm collection, preservation and utilization, in: Marshall H.G., Sorrells M.E. (Eds.), Oat Science and Technology, American Society of Agronomy, Madison Wisconsin, pp. 793–820.
- Wiberg A. (1974a) Genetical studies of spontaneous sources of resistance to powdery mildew in barley, Hereditas 77, 89–148.
- Wiberg A. (1974b) Sources of resistance to powdery mildew in barley, Hereditas 78, 1–40.
- Wieseler F., Horst W.J. (1994) Root growth and nitrate utilization of maize cultivars under field conditions, Plant Soil 163, 267–277.
- Witcombe J.R., Joshi A., Joshi K.D., Shapit B.R. (1996) Farmer participatory crop improvement I: Varietal selection and breeding methods and their impact on biodiversity, Exp. Agr. 32, 445–460.
- Wolfe M.S., Schwarzbach E. (1978) Patterns of race changes in powdery mildews, Ann. Rev. Phytopathol. 16, 159–180.
- Wright A.J., Heale J.B. (1984) Adult plant resistance to powdery mildew (*Erysiphe graminis*) in three barley cultivars, Plant Pathol. 33, 225–231.
- Wurtzel E.T. (2004) Genomics, genetics and biochemistry of maize carotenoid biosynthesis, Recent. Adv. Phytochem. 38, 85–110.
- Yahiaoui S., Igartua E., Moralejo M., Ramsay L., Molina-Cano J.L., Ciudad F.J., Lasa J.M., Gracia M.P., Casas A.M. (2008) Patterns of genetic and eco-geographical diversity in Spanish barleys, Theor. Appl. Genet. 116, 271–282.
- Yahyaoui A.H., Sharp E.L., Reinhold M. (1988) New sources of resistance to *Puccinia hordei* in barley landrace cultivars, Phytopathology 78, 905–908.
- Yasuda S., Rikiishi K. (1997) Screening of the World Barley Collection for resistance to barley yellow mosaic virus, Barley Genet. Newsl. 28, 64–66.

- Yau S.-K. (2002) Comparison of European with West Asian and North African winter barleys in tolerance to boron toxicity, Euphytica 123, 307–314.
- Yau S.-K., Nachit M.M., Ryan J., Hamblin J. (1995) Phenotypic variation in boron-toxicity tolerance at seedling stage in durum wheat (*Triticum durum*), Euphytica 83, 185–191.
- Yitbarek S., Berhane L., Fikadu A., VanLeur J.A.G., Grando S., Ceccarelli S. (1998) Variation in Ethiopian barley landrace populations for resistance to barley leaf scald and net blotch, Plant Breeding 117, 419–423.
- Zeven A.C. (1996) Results of activities to maintain landraces and other material in some European countries in situ before 1945 and what we may learn from them, Genet. Resour. Crop Ev. 43, 337–341.
- Zeven A.C. (1998) Landraces: a review of definitions and classifications, Euphytica 104, 127–139.
- Zeven A.C. (1999) The traditional inexplicable replacement of seed and seed ware of landraces and cultivars: a review, Euphytica 110, 181–191.
- Zeven A.C. (2000) Traditional maintenance breeding of landraces: 1. Data by crop, Euphytica 116, 65–85.
- Zhang X., Jin Y., Rudd R., Hall T., Rudd J., Bockelman H. (2000) Fusarium head blight resistant sources of spring wheat identified from the USDA collection. National Fusarium Head Blight Forum, Kinko's, Okemos, MI.
- Zhang Z.J. (1995) Evidence of durable resistance in nine Chinese land races and one Italian cultivar of *Triticum aestivum* to *Puccinia striiformis*, Eur. J. Plant Pathol. 101, 405–409.
- Zhang P., Dreisigacker S., Buerkert A., Alkhanjari S., Melchinger A.E., Warburton M.L. (2006) Genetic diversity and relationships of wheat landraces from Oman investigated with SSR markers, Genet. Resour. Crop Ev. 56, 1351–1360.
- Zhou L.-L., Bai G.-H., Carver B., Zhang D.-D. (2007) Identification of new sources of aluminum resistance in wheat, Plant Soil 297, 105–118.
- Zohary D., Hopf M. (1988) Domestication of Plants in the Old World, Clarendon, Oxford.
- Zohary D., Hopf M. (2000) Domestication of plants in the Old World, New York: Oxford University Press, 3rd ed, 316 p.

# **Mineral Sources of Potassium for Plant Nutrition**

David A.C. Manning

## Contents

1	Intro	oduction	188
2	Mine	eral Sources of K	191
3	The Stab	Potassium Aluminium Silicate Minerals: ility and Dissolution Rates	192
4	Pota	ssium in Silicate Rocks	193
5	Evid	ence of Silicate Minerals as Sources of K	194
6	Use Ferti	of Potassium-Bearing Silicate Rocks as a illiser	195
7	Crop	o Trials with Potassium Silicate Minerals	195
	7.1	Trials with Granite and Related Rocks	195
	7.2	Trials with Nepheline-Bearing Rocks and	
		Mineral Residues	197
	7.3	Trials with Feldspars	198
	7.4	Trials with Non-Granitic Bulk Rocks	199
8	Con	clusion	201
Re	eferen	<b>ces</b>	201

D.A.C. Manning (⊠)

School of Civil Engineering and Geosciences, Newcastle University, Newcastle upon Tyne, NE1 7RU, UK e-mail: David.Manning@newcastle.ac.uk Abstract Recently published assessments of nutrient budgets on a national basis have shown that K deficits for developing countries are so substantial that a doubling of world production of potash fertilisers would be required to balance inputs and offtake, simply to meet demands in Africa alone. The price of potassium fertiliser raw materials has increased by a factor of 4 during 2007-2009, approaching \$1000 per tonne in some markets. Thus an annual investment of the order of US\$5600 million is required to replenish soil K stocks in Africa. In this context it is appropriate to review current knowledge of alternative sources of K, which is the seventh most abundant element in the Earth's continental crust, present in feldspars and (much less commonly) feldspathoid minerals including nepheline and leucite. Theoretical considerations based on the experimental determination of mineral dissolution rates indicate that nepheline dissolves 100 times more quickly than potassium feldspar, and this suggests that nepheline-bearing rocks are more effective as sources of K for plant growth than granitic rocks, even though these have higher K contents. Crop trials with silicate rocks and minerals as sources of K show increased K availability and uptake for nepheline-bearing rocks compared with granitic rocks. Under conditions where soils are rapidly leached (especially tropical soils such as oxisols that contain quartz, aluminium oxy-hydroxides and kaolinite), with low capacity to retain soluble nutrients, the use of potassium feldspar or crushed granite does give a yield response, although no greater than for conventional fertilisers. In other experiments with crushed ultramafic, basaltic and andesitic rocks improvements in crop yield are claimed, although this cannot be unambiguously related to the mineralogical or chemical

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_11, © Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Manning, Agron. Sustain. Dev. 30 (2010) 281–294. DOI: 10.1051/agro:2009023 composition of the rock used. In conclusion, the present high cost of conventional potassium fertilisers justifies further investigation of potassium silicate minerals and their host rocks (which in some cases include basic rocks, such as basalt) as alternative sources of K, especially for systems with highly weathered soils that lack a significant cation exchange capacity. Such soils commonly occur in developing countries, and so this approach provides an opportunity to develop indigenous silicate rock sources of K as an alternative to sometimes prohibitively expensive commercial fertilisers.

## 1 Introduction

Of the three main plant nutrients, N, P and K, potassium (K) and phosphorus (P) are exclusively sourced from geological materials (Manning, 1995; van Straaten, 2007). Both nutrients are mined and processed to give fertiliser products that vary in the amount of chemical treatment involved in their preparation. Both can be used as mined (sylvinite (mixed KCl + NaCl) and phosphate rock). However, the majority of commercial phosphate fertilisers are manufactured from phosphoric acid, including composite N-P fertilisers, such as diammonium phosphate (DAP), which incorporate N as urea, ammonium or nitrate. The manufacture of nitrogen fertilisers depends largely on chemical processes that derive N from atmospheric sources or from fossil fuel materials, with subsequent energy-intensive processing to give a final product. Per unit of nutrient, N fertiliser production typically uses an order of magnitude more energy than required to produce P or K fertilisers (Lægrid et al., 1999).

World trade in fertilisers is substantial: in 2007–8, consumption is reported to be 98 million tonnes N, 40 million tonnes P (as  $P_2O_5$ ), and 28 million tonnes K (as  $K_2O$ : FAO, 2008). However, trade does not necessarily reflect agricultural need, but ability to purchase. This can be demonstrated using nutrient budgets on a national and international scale, one based on trade (e.g. FAO, 2008), and the other based on scaling up farm offtakes and inputs (e.g. Sheldrick et al., 2002).

In terms of the global trade in fertiliser products, the Food and Agriculture Organisation of the United Nations (FAO, 2008) indicates that global supply of fertilisers is sufficient to meet demand. For potassium, the FAO states that global demand is likely to increase annually at 2.4%, and that supply will balance demand. However, in an analysis of the regional situation, the FAO notes that only 10 of 57 African countries consume significant quantities of fertiliser of any kind. Consumption in Africa of K fertilisers is currently approximately 485 000 tonnes (2008–9), expected to increase by 2% annually, all of which has to be imported (FAO, 2008). Other regions that rely on imported K include Oceania and Asia, with the bulk of commercial supply derived from Europe and North America.

In an assessment of nutrient budgets, Sheldrick et al. (2002) describe a nutrient audit model, in which offtakes are expressed in terms of nutrients removed from the land, and inputs include fertilisers, crop residues and manures. These are then balanced and assessed on a national, regional or global scale to determine whether or not nutrient supply is balanced by inputs. Importantly, Sheldrick et al.'s (2002) work has shown that on a global basis it is not phosphorus supply that is of most concern, but supply of potassium, with an annual global deficit of 20 kg K ha<sup>-1</sup>. In an assessment of nutrient audits for Africa, Sheldrick and Lingard (2004) have shown that overall, and in most African countries, nutrient depletion has increased since 1961, with annual deficits in 1998 of 3.5 million tonnes of N (17.4 kg N ha<sup>-1</sup> year<sup>-1</sup>), 0.7 million tonnes of P  $(3.3 \text{ kgPha}^{-1} \text{ year}^{-1})$  and 4.1 million tonnes of K  $(20.0 \text{ kg K ha}^{-1} \text{ vear}^{-1})$ . The highest nutrient depletion rates are for potassium.

Using data from Sheldrick and Lingard (2004) for African countries, Fig. 1 shows the extent of



**Fig. 1** Potassium inputs and outputs combined for all African countries for the latter part of the 20th Century (Sheldrick and Lingard, 2004)

the imbalance between the supply and offtake of K. Between 1961 and 1998, the total output of K through offtakes has almost doubled, to 30 kg K ha<sup>-1</sup> year<sup>-1</sup>. In contrast, fertiliser K input has remained very low, at 2.1 kg K ha<sup>-1</sup> year<sup>-1</sup> or less, whilst other inputs (from crop residues and manures) have increased. Potassium fertilisers make up 10% or less of the total K input, and this proportion has declined between 1961 and 1998.

The nutrient deficits identified by Sheldrick and Lingard (2004) and the problems of fertiliser supply identified by the FAO (FAO, 2008) for Africa have major implications for the supply of potash to agriculture, more widely within the developing world. This is because potash ores have a rather limited distribution globally (Rittenhouse, 1979; Moores, 2009b), with the bulk of the world's potash mined in Canada, Europe and the Middle East. Thus there is currently very little scope for many developing countries to be self-sufficient in potash using conventional fertilisers.

Furthermore, consideration of trade in the context of geographical factors, infrastructure and transport costs confirms, for example, that many (but by no means all) African countries do not compete on a level playing field with more developed countries. Limão and Venables (2001) carried out a detailed analysis using gravity modelling to assess the impact of a number of variables on the costs of trade. For sub-Saharan African countries, transport costs between countries are 136% higher than for trade between other African countries, and infrastructure costs account for half of this figure. Poor infrastructure plays a highly significant part in inhibiting trade within Africa, including transit from coastal ports to landlocked countries (Longo and Sekkat, 2004; Limão and Venables, 2001).

Focusing on potash fertilisers, and considering the situation in Africa, Fig. 2 shows the amount of muriate of potash that would be needed to compensate for potassium offtake for the period shown in Fig. 1. Figure 2 shows that in 1961 inputs from fertilisers equivalent to 3 million tonnes of KCl were required to balance potassium offtake, rising to 7 million tonnes annually for the period 1995–1998. This is 70% of the annual consumption of muriate of potash by the United States during the same period (10 million tonnes per annum; Searls, 2000), and about 20% of world consumption at that time. Assuming that demand for K in Africa has not increased since 1998, the value of the annual fertiliser need for Africa identified in this way is of the order of US\$5600 million using prices for the



**Fig. 2** Estimated total requirement of K for African countries, based on difference between outputs and existing inputs

end of 2008, likely to be an underestimate given that demand is expected to have increased since 2000.

The situation described above concerns an entire diverse continent, and varies from one African country to another. But similar deficits occur elsewhere: for 1998, China has a K deficit of 60 kg K ha<sup>-1</sup> year<sup>-1</sup>, or a requirement of 8.3 million tonnes per year. To maintain this deficit and to feed its growing population, Sheldrick et al. (2003) consider that China needs to see growth in K fertiliser use at an annual rate of 8%. According to the FAO, East Asia's K deficit is in excess of 9 million tonnes per year, dominated by China (FAO, 2008). In contrast to the developing world, the USA and Europe have no potash deficit; in general, developed countries add K well in excess of offtake.

Using figures from Sheldrick et al. (2002) for the end of the 20th century, globally, there is a deficit of 20 kg K ha<sup>-1</sup> year<sup>-1</sup>, corresponding to an annual global fertiliser need of 30 million tonnes of K. This corresponds to a requirement for an additional 60 million tonnes of KCl (35 million tonnes of K<sub>2</sub>O equivalent), which is slightly greater in quantity than the most recent published estimates of current world production (approx. 30 million tonnes of K<sub>2</sub>O equivalent; Ober, 2007).

Thus to satisfy demand from world crop production for K, the amount of additional potash required is equivalent to slightly more than current global mined production. Production and consumption of potash fertilisers must more than double to sustain soil K stocks. At present, the potassium deficit that arises from offtake of crops is being supplied by the soils. Using K as an indicator of other mineral-derived nutrients, it



**Fig. 3** Changes in prices to July 2009 of muriate of potash (MOP), urea and diammonium phosphate (DAP) fertilizers, relative to January 2007 prices (calculated using data from ICIS; http://www.icis.com)

appears that global agricultural production is responsible for nutrient mining on a huge and evidently unsustainable scale.

The above review of Sheldrick's work is based on nutrient balances for the late 20th Century. The first decade of the 21st Century has seen the growth of China as an industrial power, with concomitant demand for natural resources, including fertilisers. This has led to substantial demand for K, and world prices have risen accordingly.

Figure 3 compares changes in price (not corrected for inflation) for three fertilisers since the start of 2007: urea, diammonium phosphate (DAP) and muriate of potash (MOP). For N and P fertilisers, it clearly shows an increase and subsequent decrease that coincides broadly with trends in oil prices that occurred in the latter part of 2008, reflecting the close links between these and the fossil fuel industry. Prices for these fertilisers return to values similar to those of January 2007 (DAP; approximately US\$ 280/tonne compared with US\$ 250/tonne in January 2007) or are lower (urea; approximately US\$ 280/tonne compared with US\$ 350/tonne in January 2007). In contrast, the price of potash (muriate of potash, price Freight On Board (FOB) Saskatchewan) for the period rose by a factor of 4 in this period, from US\$180/tonne to US\$650/tonne, with a peak price of US\$ 800/tonne for much of late 2008, and spot market prices of US\$1000/tonne reported on the internet. Most importantly, the price of K has not decreased in parallel with the observed decrease in price of N and P fertilisers. This observation

is discussed in more detail by Moores (2009a, b), and fundamentally distinguishes potash fertilisers. The high price of potash reflects factors in addition to those associated with manufacture, including inability of supply to keep pace with demand.

Given the apparent obstacles to trade and the high cost of potash, it is important to consider how the African potash deficit identified by Sheldrick and Lingard (2004) can be addressed. One possibility is to identify new sources of K. There are known deposits of potash salts (e.g. Republic of Congo, De Ruiter, 1979; Thailand, El Tabakh et al., 1999) that presently are not mined. The Republic of Congo deposits (located 16 km east of the port of Pointe-Noire) are (according to internet sources) at an advanced stage of mine planning and may represent a significant source in due course. However, in general the distribution of conventional mined potash salts has changed little since Rittenhouse (1979) considered global supply and demand, and is unlikely to change given the very substantial cost of developing a producing mine (US\$2000 million; Moores, 2009b). In any case, given the difficulties of intra-African trade (Limão and Venables, 2001), the possible development of the Pointe-Noire potash deposits is unlikely to improve the availability of K in those countries that most need it, especially as the products of the mine are likely to be sold on the world market at a similar price to those from other sources.

In these circumstances it is appropriate to consider the availability of K within soil systems from first principles, and to consider unconventional sources of K that are not necessarily attractive on a commercial basis to the global fertiliser industry but that might be appropriate in circumstances where farmers are presently excluded from global fertiliser markets.

In soil systems, retention of potassium is wholly dependent on the cation exchange capacity of the soil. This, in turn, is largely controlled by the presence within soils of predominantly clay minerals that have an interlayer site capable of accommodating K, such as the illite group or vermiculites (Sposito, 1989). In two of the major soil orders, oxisols and ultisols, cation exchange capacity is very low, reflecting the leaching from these of cations as a consequence of the extreme weathering process that they have undergone. Such soils are characterized by the presence of quartz, kaolinite and Fe-Al oxy-hydroxides, representing a residual mineral assemblage from which nutrients such as K have been leached. These soils occur widely in tropical

Table 1         Chemical formulae	Mineral	Formula	Weight % K	Weight % K <sub>2</sub> O				
and potassium contents (expressed as element and	Potash ore minerals							
oxide) for potash ore minerals	Sylvite	KCl	52.35	63.09				
and for common potassium	Carnallite	MgCl <sub>2</sub> .KCl.6H <sub>2</sub> O	14.05	16.94				
silicate rock forming minerals	Kainite	KMgSO <sub>4</sub> Cl.3H <sub>2</sub> O	15.69	18.91				
	Langbeinite	2MgSO <sub>4</sub> .K <sub>2</sub> SO <sub>4</sub>	18.84	22.71				
	Silicate minerals	Silicate minerals						
	Potassium feldspar	KAlSi <sub>3</sub> O <sub>8</sub>	14.03	16.91				
	Leucite	KAlSi <sub>2</sub> O <sub>6</sub>	17.89	21.56				
	Nepheline	(Na,K)AlSiO <sub>4</sub>	13.00	15.67				
	Kalsilite	KAlSiO <sub>4</sub>	24.68	29.75				
	Muscovite	KAl <sub>3</sub> Si <sub>3</sub> O <sub>10</sub> (OH) <sub>2</sub>	9.03	10.88				
	Biotite	$K_2Fe_6Si_6Al_2O_{20}(OH)_4$	7.62	9.18				
	Phlogopite	K2Mg6Si6Al2O20(OH)4	9.38	11.30				

areas of Africa, South America and Australia. Their cultivation requires addition of nutrients including K, which are removed by through drainage as well as by plant uptake.

For an agricultural system in a poor country in which oxisols dominate, the supply of K is a critical limiting factor. In this context it is appropriate to consider unconventional geological sources of K that, whilst less effective than the soluble salts that make up commercial fertiliser products, are widely available globally and might be expected to weather rapidly within a soil, releasing sufficient K to provide agronomic benefit. Such sources include silicate rocks, whose use as a source of plant nutrients has been discussed by a number of authors (van Straaten, 2006, 2007; Leonardos et al., 1987; Harley and Gilkes, 2000; Gillman et al., 2002).

The purpose of this review is first to present the theoretical background that justifies the use of crushed silicate rocks as a source of plant nutrients, focusing on K, and secondly to provide an overview of the results of existing published trials using crushed rocks as an alternative to conventional fertilisers. It then considers in principle whether or not crushed silicate rocks should be considered as one source of K that can address deficits on a basis that is appropriate for sustainable crop production, and highlights the need for strategic and consistent future trials.

## 2 Mineral Sources of K

In the context of meeting the world's need for potash fertilisers, there is no doubt that conventional soluble salts will continue to dominate in developed countries. Table 1 lists the principle potash fertiliser ore minerals, together with the dominant rock-forming potassium silicate minerals. In natural soils, potassium is derived ultimately from the potassium silicate minerals, which can be subdivided into those that are formed at high temperatures by igneous and metamorphic processes (feldspars and feldspathoids; primary micas), and those that are produced on weathering of the hightemperature minerals (clay minerals, especially illite). In deeply weathered, highly oxidized soils that have low organic matter, it is essentially the clay minerals that are responsible for the exchangeable K that is measured in soil analysis; the high temperature silicate minerals provide a reservoir of fixed K.

It is normally assumed that the weathering of the feldspars and feldspathoids is very slow, and that their addition to soil will not be beneficial for crop growth, especially when compared with conventional potash fertilisers, which, being salts, are readily soluble as well as having high K contents. A number of studies have investigated silicate rocks as sources of K (and other nutrients), in response to the following drivers:

- nutrient supply for organic farmers, principally in developed countries, who seek an alternative to salts (partly in response to environmental protection demands). This is in part related to the 'Remineralise the Earth' philosophy of plant nutrition using rock dust (http://remineralize.org);
- (2) nutrient supply for farmers in developing countries who lack funds to purchase conventional fertilisers.

Both areas of application have shown that in some circumstances potassium silicate rocks can act as effective sources of K, with long term crop yields that meet the needs of the producer. A number of commercial enterprises use potassium silicate rocks in long-standing businesses with satisfied customers, particularly in the organic farming sector (e.g. http://www. glensideorganics.co.uk). However, the scientific evidence for the possible agronomic value of potassium silicate rocks is dispersed in a number of publications, and these do not generally refer to specific commercial products.

# 3 The Potassium Aluminium Silicate Minerals: Stability and Dissolution Rates

Potassium occurs as an essential component within the feldpars and felspathoid mineral groups. In both groups, the mineral structure is a three dimensional network of silica tetrahedra, in which substitution of Al for Si leads to a charge deficiency that is balanced by the presence of Na<sup>+</sup>, K<sup>+</sup> or Ca<sup>++</sup>. The potassic aluminosilicate minerals and their formulae are summarised in Table 1, which also gives their theoretical K<sub>2</sub>O contents, ranging from 18-30 wt% (equivalent to 15–25 wt%; 150–250 g kg<sup>-1</sup> K). Of the aluminosilicate minerals, potassium feldspar is most commonly found, occurring in a wide range of rock types (granite, sandstone, gneisses, etc). Orthoclase (Or) and microcline (Mc) are common varieties of potassium feldspar, differing in their crystal structure (Deer et al., 1992). Leucite (Lc), nepheline (Ne) and kalsilite (Ks) are much less widely distributed, predominantly occurring in nepheline syenites and related volcanic rocks, and some alkaline basalts. Nepheline is relatively readily available as a commodity, as it is mined as a raw material for ceramic and glass manufacture. End-member nepheline is a sodium aluminium silicate; natural nepheline typically contains potassium to the extent of 20% of the cation site. Leucite and

kalsilite are not readily available as commodities, and often occur mixed with secondary products of their weathering and alteration.

The thermodynamic stability of the potassium aluminium silicates can be assessed using the approach of Curtis (1976). A balanced chemical reaction can be written in which aluminium is conserved and the weathering products include mineral phases that are known to be stable under Earth-surface conditions. This approach provides a thermodynamic basis for empirical stability series such as that produced by Goldich (1938; Curtis, 1976). Table 2 shows the reactions for the aluminosilicates, including the plagioclase feldspars anorthite (An) and albite (Ab) for comparison. The calculated free energy changes for the reactions are given in Table 2, taking into account variation in the written formula by expressing the free energy change per gram atom rather than per mole (this approach allows comparison of like with like in terms of the numbers of bonds being broken as the framework silicate weathers; Curtis, 1976). The least stable mineral is kalsilite, followed by nepheline, then leucite and finally potassium feldspar (microcline). On the basis of these thermodynamic data, kalsilite and nepheline stabilities are similar to those of the calcic plagioclase feldspars.

In addition to variation in their thermodynamic stability, the potassium aluminosilicates vary in the kinetics of their dissolution reactions. Data for these are available for the feldspars (summarised by Blum and Stillings, 1995) and for nepheline (Tole et al., 1986); there are no comparable dissolution rate data for leucite or kalsilite. Dissolution of the aluminosilicates takes place as a consequence of hydrolysis reactions at the mineral surface (Brantley and Stillings, 1996); these reactions are irreversible. They liberate cations to the solution and disrupt the framework structure of the mineral. Thus they differ fundamentally from the reversible cation exchange reactions that characterise the interlayer site of the clays and micas and the exchange site of the zeolites.

 Table 2
 Weathering reactions for framework aluminosilicate minerals (Curtis, 1976)

	6	· · · ·	
	Reaction	$\Delta G_r$ , kJ/mol	$\Delta G_r$ , kJ/ gram atom
Ks	2KAlSiO <sub>4</sub> + 2H <sup>+</sup> + H <sub>2</sub> O = 2K <sup>+</sup> + Al <sub>2</sub> Si <sub>2</sub> O <sub>5</sub> (OH) <sub>4</sub>	-115.25	-6.07
An	$CaAl_2Si_2O_8 + 2H^+ + H_2O = Ca^{2+} + Al_2Si_2O_5(OH)_4$	-98.49	-5.47
Ne	$2\mathrm{NaAlSiO}_4 + 2\mathrm{H}^+ + \mathrm{H}_2\mathrm{O} = 2\mathrm{Na}^+ + \mathrm{Al}_2\mathrm{Si}_2\mathrm{O}_5(\mathrm{OH})_4$	-91.02	-4.79
Lc	$2KAlSi_2O_6 + 2H^+ + H_2O = 2K^+ + Al_2Si_2O_5(OH)_4 + 2SiO_2$	-88.00	-3.52
Ab	$2NaAlSi_3O_8 + 2H^+ + H_2O = 2Na^+ + Al_2Si_2O_5(OH)_4 + 4SiO_2$	-87.74	-2.83
Mc	2KAlSi <sub>3</sub> O <sub>8</sub> + 2H <sup>+</sup> + H <sub>2</sub> O = 2K <sup>+</sup> + Al <sub>2</sub> Si <sub>2</sub> O <sub>5</sub> (OH) <sub>4</sub> + 4SiO <sub>2</sub>	-67.70	-2.18

I I I I I I I I I I I I I I I I I I I		-,,
	Dissolution rate	Relative
Mineral	Log k+	dissolution rate
Potassium feldspar	-9.93 (pH 1-5.7)	1.9
	-10.2 (basic)	1
Leucite	Not known	_
Nepheline	-8.2 (pH5)	100
	-8.6 (pH7)	40
Kalsilite	Not known	_
Anorthite	-5.87	21000
Albite	-9.69	3.2

**Table 3** Dissolution rates for potassium silicate minerals and other feldspars (date from Blum and Stillings, 1995)

The available dissolution rate data for the feldspars and for nepheline are given in Table 3. It is important to compare corresponding experimental conditions, as a wide range of temperatures, pH and other factors has been considered in kinetic studies. Blum and Stillings' review (1995) assembles directly comparable information in one source. It is also important to note that these experimentally-determined dissolution rates are not directly applicable to the soil environment (e.g. White, 1995). Of particular importance is the experimental difficulty in determining the reacting surface area of the mineral. The feldspars (and presumably the feldspathoids) characteristically are riddled with holes and structural defects at a very fine (molecular) scale (Lee and Parsons, 1995), and these can have a major effect on dissolution kinetics.

Table 3 also shows relative dissolution rate data for the potassium aluminosilicates, albite and anorthite. Dissolution rates vary according to pH; for comparable pH values, nepheline's dissolution rate exceeds that of potassium feldspar by as much as 100 times. Because of its lower thermodynamic stability compared with nepheline, it might be predicted that kalsilite has a higher dissolution rate, although its similar framework structure means that its dissolution rate is likely to be similar to that of nepheline. Leucite is expected to have a dissolution rate that is intermediate between nepheline and potassium feldspar.

## 4 Potassium in Silicate Rocks

The potassium silicate minerals typically occur mixed with other silicate minerals, and this dilutes the K content. In general terms, K is the seventh most abundant element in the Earth's continental crust, the composition of which is dominated by rocks of granitic composition and the products of their weathering and erosion. Potassium feldspar is an essential component of granitic rocks, which also contain the micas biotite and muscovite. Weathering of granitic rocks initially produces secondary muscovite and chemicallysimilar illitic clays, and if conditions are right ultimately produces the potassium-free clay mineral kaolinite.

Although potassium feldspar, muscovite and biotite are very common, the other potassium silicate minerals are rare. Nepheline occurs in very specific igneous rock types that occur sporadically, mainly as intrusive rocks but also within lavas. Kalsilite and leucite both occur mainly in volcanic rocks (Deer et al., 1992), in potassium-rich lavas from specific regions. These include rift valleys, such as the East African Rift (Woolley, 2001).

Examples of typical rock compositions are given in Table 4. This table also shows estimated compositions for the upper continental crust, based on the observed composition of till derived from glacial erosion of (a) igneous rocks and (b) sedimentary (Carboniferous) sequences. Table 4 also gives the composition of particulates carried by the Ganges, which contribute nutrients to soils when deposited in annual floods. Table 4 shows that the potassium content of some readily available, and some rare, rock materials are quite similar, ranging from  $2.7-5.6 \text{ wt}\% \text{ K}_2\text{O}$ .

Table 4 also recalculates the composition of the rock or sediment in terms of the equivalent hypothetical, normative, content of orthoclase (K-feldspar), nepheline and leucite. The normative composition shows that glacial sediments derived from granitic terrains are chemically very similar (in terms of K content) to their source, justifying consideration of the use of tills of this type for soil remineralisation. However, glacial till can be highly variable in composition (Barlow, 1996), its composition reflecting mixing between different mineral components within the substrate. Thus the samples reported by Barlow (1996) show mixing between quartz (derived from sandstone) and a clay-rich fraction derived from mudstones. Tills derived from sedimentary rocks may have lower K contents, and lower normative orthoclase, as a consequence of weathering prior to till formation.

	Granite	Trachyte	Diorite	Nepheline syenite	Basalt	Nephelinite	Upper continental crust (glacial sediments)	Glacial till (clay-rich)	Ganges river particulates
Comments:	Common	Rare	Moderately common	Rare	Common	Very rare	Igneous/ Metamorphic crust	Sedimentary rock source	
Oxide									
SiO <sub>2</sub>	71.30	61.21	57.48	54.99	49.20	40.60	62.22	79.31	66.9
TiO <sub>2</sub>	0.31	0.70	0.95	0.60	1.84	2.66	0.83	0.39	0.96
$Al_2O_3$	14.32	16.96	16.67	20.96	15.74	14.33	16.63	7.54	16.0
Fe <sub>2</sub> O <sub>3</sub>	1.21	2.99	2.50	2.25	3.79	5.48	_		_
FeO	1.64	2.29	4.92	2.05	7.13	6.17	6.99*	3.21*	5.2*
MnO	0.05	0.15	0.12	0.15	0.20	0.26	0.12	0.05	0.14
MgO	0.71	0.93	3.71	0.77	6.73	6.39	3.47	1.11	2.3
CaO	1.84	2.34	6.58	2.31	9.47	11.89	3.23	1.16	4.1
Na <sub>2</sub> O	3.68	5.47	3.54	8.23	2.91	4.79	2.15	0.57	1.5
K <sub>2</sub> O	4.07	4.98	1.76	5.58	1.10	3.46	4.13	1.74	2.7
$P_2O_5$	0.12	0.21	0.29	0.13	0.35	1.07	0.23	0.09	—
Normative potassic minerals									
Or	24.50	29.41	10.42	32.98	6.53	3.16	24.41	10.28	15.96
Lc	_	_	_	_	_	13.57	_	—	—
Ne	_	_	_	21.77	_	21.95	_	_	_

**Table 4** Typical compositions of igneous rocks (Le Maitre, 1976) and estimates for continental crustal composition based on till and suspended sediment composition (Goldschmidt, 1933; Rudnick and Gao, 2003; Martin and Meybeck, 1979)

\*Total iron reported as FeO

# 5 Evidence of Silicate Minerals as Sources of K

A number of studies have addressed the ability of different potassium silicate minerals to yield nutrients under laboratory and pot trial conditions.

The ability of feldspar to dissolve in soil conditions, yielding K, has been determined by Riggs et al. (1993). Potassium feldspar was added to soil and subjected to leaching and incubation experiments. These showed that feldspar leaching was effective in open system leaching experiments, more so that in closed-system incubations. Similarly, Blum et al. (1989) compare a number of different rock types as sources of mineral plant nutrients. Assuming that micas and feldspars are concentrated in the silt and sand fractions of a soil, Mengel et al. (1998) compared the extractable K contents and the performance of ryegrass (Lolium perenne) for whole soil and for the same soil from which the clay fraction had been removed. Finding no difference between the two soils, Mengel et al. (1998) conclude that K is as easily available from the sand and silt fraction as from the whole soil. In earlier trials with soils that differ in their sand, silt and clay content (again with no detailed mineralogical or geochemical analysis), Mengel and Rahmatullah (1994) observed potassium deficiency in wheat (*Triticum aestivum* cv. Faisalabad 83), maize (*Zea mays* L. cv Gohar) and barley (*Hordeum vulgare*), but adequate K nutrition of elephant grass (*Pennisetum purpureum*).

The interaction between silicate minerals and microbial systems has been investigated by a number of authors. In the context of fungal community development on natural weathered rock surfaces, Gleeson et al. (2005) show that individual minerals are associated with distinct fungal community structures that relate to the chemical composition of the mineral. Similarly, Rosling et al. (2004) showed that eight different ectomycorrhizal fungal cultures responded differently to culturing in the presence of minerals (quartz, apatite, potassium feldspar and marble).

In laboratory experiments in which the fungus *Aspergillus fumigatus* was applied to a potassium-rich shale (containing principally K-feldspar), K release was enhanced by two orders of magnitude (Lian et al., 2008). In addition to the role played by organic acids exuded by the fungus, Lian et al. (2008) observed increased K yields when the organism was in contact with the mineral substrate, suggesting that surface interactions, some of which are mechanical, enhance K release.

Wallander and Wickman (1999) compared ectomycorrhizal and non-mycorrhizal cultivation of *Pinus sylvestris* seedlings in pot trials where the source of K was potassium feldspar (microcline) or biotite. In both cases, release of K from the minerals (as measured by biomass K contents) was enhanced by the presence of mycorrhizal fungi.

# 6 Use of Potassium-Bearing Silicate Rocks as a Fertiliser

Proposals to use potassium silicate rocks and minerals as sources of plant nutrients are not new; Lloyd (1918) comments that it is economically not viable to extract K from silicate rocks to produce fertilisers, despite the availability of a number of different methods. Instead of regarding potassium silicate-bearing rocks as an 'ore' from which to extract K artificially, it is more appropriate to consider their direct use, allowing plants and their root processes to extract the K that they need.

Selection of a silicate rock potash fertiliser can be based on the absolute potash content, which is then tested for availability (e.g. Barral Silva et al., 2005). An alternative approach is to select on the basis of dissolution rate. The value of this approach is illustrated in Fig. 4, which shows the relative rates of loss of K to solution from two silicate minerals, potassium feldspar and nepheline, at neutral pH and pH 5 over a period of 1 year, for a surface area of  $1 \text{ m}^2 \text{ g}^{-1}$ . Figure 4 clearly shows that the release of K from nepheline, with 3.8% K, is an order of magnitude greater than release from potassium feldspar (15% K), and that this observation applies to pH values that are appropriate for manures and soil solutions influenced by active root growth. Thus rocks that contain nepheline (and by implication kalsilite and leucite) are likely to be more effective as sources of K than those that contain K-feldspar alone.

# 7 Crop Trials with Potassium Silicate Minerals

Table 5 summarises a number of trials that have used silicate rock materials as a source of plant nutrients, focusing on K. These trials include pot (greenhouse) and field experiments, using a range of differing crops, for differing time scales and in different climates.

# 7.1 Trials with Granite and Related Rocks

A consistent set of trials has been carried out in Western Australia, in which the same granite was used in field trials on wheat, and pot trials on wheat, clover and ryegrass (Coreonos et al., 1996; Hinsinger et al., 1996; Bolland and Baker, 2000). Although a consistent source of granite powder was used, the K content varied from one study to another, from 2.2-3.8% K<sub>2</sub>O. Application rates also varied, up to the equivalent of



Fig. 4 Release of K (expressed as milliequivalents/gram) from nepheline and potassium feldspar calculated from mineral dissolution rates

Crop	Species/cultivar	Mineral/rock	Trial type	Duration (months)	Agronomic benefits	Reference
Alfolfo	Madiagao sating L ou	Gnoise	Pot	1.75	Insignificant	Wang at al. 2000
Allalla	Asta and cy Haifei	Olleiss	FUL	1.75	Insignificant	wang et al., 2000
Clover	Trifolium subterraneum	Granite	Field/pot	12/1-1.5	Insignificant	Bolland and Baker, 2000
Clover	Trifolium subterraneum cv Trikkala	Granite	Pot	5	Increased yield and K uptake	Coreonos et al., 1996
Grass	Timothy Phleum protense L., Meadow fescue Festuca protense L.	Ksp and Ne+Bi- rich mine tailings	Field	36	K uptake similar for Ne-bearing tailings and KCl; Ksp-hosted K inadequately available.	Bakken et al., 2000
Grass	Brachiaria dyctioneura	Feldspar (Ksp)	Field	14	Insignificant	Sanz Scovino and Rowell, 1988
Grass	Italian ryegrass Lolium multiflorum L.	Gneiss	Pot	1.75	Significant K mobilization	Wang et al., 2000
Grass	Perennial ryegrass Lolium perenne L.	Gneiss	Pot	1.75	Significant K mobilization	Wang et al., 2000
Grass	Rye grass <i>Lolium</i> rigidum cv Standard	Granite	Pot	5	Increased yield and K uptake	Coreonos et al., 1996
Italian ryegrass	<i>Lolium multiflorum</i> <i>italicum</i> var . Turilo	Ksp and Ne+Bi- rich mine tailings	Pot	6	Ksp-hosted K inadequately available. Increased yield for Bi and Ne-bearing rocks.	Bakken et al., 1997
Legume	Pueraria phaseoloides	Feldspar (Ksp)	Field	14	Insignificant	Sanz Scovino and Rowell, 1988
Maize Okra	Zea mays l. cv ND60 Abelmoschus esculentus cv Eskandrani and cv Balady	Gneiss Feldspar (Ksp)	Pot Field	1.75 24	Significant K mobilization Increased pod yield compared with conventional fertiliser	Wang et al., 2000 Abdel-Mouty and El-Greadly, 2008
Onions	Allium cepa, L.	Feldspar (Ksp)	Field	24	15% less yield than equivalent chemical fertiliser	Ali and Taalab, 2008
Pak choi	Brassica campestris L. ssp. Chinensis L.	Gneiss	Pot	1.75	Significant K mobilization	Wang et al., 2000
Rice	<i>Oryza</i> sp.	Phlogopite	Pot		Increased grain yield	Weerasuriya et al., 1993
Spruce	Picea sp.	Phonolite	Field	60	Increased uptake	von Wilpert and Lukes, 2003
Tomatoes	Solanum lycopersicum: variety not stated	Feldspar (Ksp)	Field	12	Increased yield	Badr, (2006)
Wheat	Triticum aestivum	Granite	Field/pot	12/1-1.5	Insignificant	Bolland and Baker, 2000
Wheat	<i>Triticum aestivum</i> cv Gutha	Granite	Pot	2.5	Increased yield	Hinsinger et al., 1996
Wheat	<i>Triticum aestivum</i> cv Gutha	Diorite	Pot	2.5	Insignificant	Hinsinger et al., 1996

Table 5 Summary of trials with silicate rocks as nutrient sources (K). Ksp: potassium feldspar; Ne: nepheline; Bi: biotite

30 T rock per hectare (up to 1 T ha<sup>-1</sup> K). Contrasting results were obtained. For pot experiments with wheat, Hinsinger et al. (1996) used granite (2.29% K<sub>2</sub>O) and found significantly increased (by 10%–20%) biomass yields for applications of 7.5 T rock ha<sup>-1</sup> (equivalent to 0.24 t K ha<sup>-1</sup>). Experiments with ground diorite (0.3% K<sub>2</sub>O) gave no increase in yield. Other work with wheat was inconclusive: Bolland and Baker report

wheat yields for field and pot trials with applications of 2, 5 and 20 T rock per hectare (again 2.29%  $K_2O$ ), with no increase in yield compared with the control. In contrast, pot trials using clover and ryegrass with the equivalent of 20 T rock per hectare (3.85%  $K_2O$ ; Coreonos et al., 1996) showed that application of granite powder did enhance both yield and shoot K content significantly compared with control. In all cases similar sandy soils with very low exchangeable K were used. From this work, it was concluded that the benefits of using crushed rock materials as a source of K are not sufficient in view of the cost differential compared with conventional soluble K sources (Harley and Gilkes, 2000). As the price of K has risen by a factor of at least 4 since 2000 (US\$ 120/tonne), this conclusion may no longer be valid.

Pot trials with gneiss, which is a highly metamorphosed rock mineralogically similar to granite, are reported by Wang et al. (2000). The mineralogical and chemical compositions of the gneiss used in these experiments are not given, apart from the K content  $(29.6 \text{ g K kg}^{-1} \text{ or } 3.57\% \text{ K}_2\text{O})$ , and the observation that feldspar, muscovite and biotite are present. The purpose of the experiment was to compare the ability of different plants to extract K from the minerals within the gneiss, using maize (Zea mays L. cv. ND60), pak choi (Brassica campestris L. ssp. chinensis), Italian ryegrass (Lolium multiflorum L.), perennial ryegrass (Lolium perenne L.) and two alfalfa cultivars (Medicago sativa L. cv. Asta and cv. Haifei). Dry yields were found to be similar for plants grown with gneiss compared with those supplied with a corresponding nutrient solution, but K uptake was lower for plants grown with gneiss. Comparing different grain sizes for the crushed gneiss, K uptake was greatest for finer size fractions of gneiss, and in all cases greatest for maize, then ryegrass and pak choi. Wang et al. (2000) conclude that plant growth facilitates the release of K from gneiss, which contains the potassium minerals feldspar, muscovite and biotite, with varying efficiency that relates to root biomass and root activity.

# 7.2 Trials with Nepheline-Bearing Rocks and Mineral Residues

To compare the possible effects of different potassiumbearing silicate minerals, studies reported by Bakken et al. (1997 and 2000) present data for experiments involving (a) Italian ryegrass (*Lolium multiflorum italicum* var. Turilo) in pot trials (Bakken et al., 1997) and (b) the growth of a mixture of timothy (*Phleum pratense* L.) and meadow fescue (*Festuca pratensis* L.) in field trials in central and northern Norway (Bakken et al., 2000). The sources of potash were mine tailings (nepheline-bearing carbonatites (an igneous carbonate rock); nepheline syenites), a quarried epidote schist, and commercial products (adularia shale, biotite concentrate). All K sources, including KCl for comparison, were applied at the rate of  $50 \text{ kg K ha}^{-1}$  in the first and second years only. Over a three year period, dry yield and K offtake were determined.

In their analysis of the results from these trials, Bakken et al. (2000) note that dry yields with KCl application were greatly in excess of those for the rock applications for years 1 and 2, but insignificantly different for year 3. Treatments with feldspar-rich rocks did not differ significantly from control, but those with biotite and nepheline-bearing rocks did show increased dry yields. However, in pot trials, Bakken et al. (1997) note that K uptake is similar for applications of nepheline-bearing carbonatite fines and KCl, and that results for experiments with K feldspar gave similar uptake to controls.

Figure 5 illustrates the different performance of the minerals used in Bakken et al.'s (2000) trials. In Fig. 5, the relative contributions of potassium feldspar, nepheline and biotite to the potassium content of the rock are expressed as an index (K-feldspar index =%K-feldspar/(%K-feldspar + %nepheline + %biotite), etc.) that allows the different rock types to be compared. These figures show that in general there is an increase in yield, compared with control, as the proportions of both nepheline and biotite increase. The zero nepheline index samples include the control and results for trials with 100% biotite. Because these figures concern relative proportions of the three minerals, the apparent decline in yield with increasing K-feldspar index (Fig. 5a) reflects the decreasing proportion of either nepheline or biotite.

A related trial with potentially nepheline-bearing rock is reported by von Wilpert and Lukes (2003), who used phonolite from the German Kaiserstuhl carbonatite complex in experiments with spruce (*Picea* sp.) on a glacial loam. Although the detailed mineralogy and origin of the phonolite is not given, in the context of published descriptions of phonolites from this location it can be assumed that if correctly named it contains nepheline (a characteristic of this rock type). von Wilpert and Lukes (2003) report that the phonolite used in their trials contains 4.23% K (5.1% K<sub>2</sub>O), and that the rock may have contained zeolite minerals (giving it enhanced cation exchange capacity). The trials compared dolomite, phonolite and potassium sulphate, and showed that after 5 years the application



Fig. 5 Comparison of the performance of rock types used by Bakken et al. (2000) on the basis of their relative proportions of potassium feldspar, nepheline and biotite

of the two rock powders had influenced soil properties measurably, with an increase in the exchangeable ion pool. Nutritional effects were determined from the K content of the spruce needles, which after 4 years increased with the application of phonolite (but still remained below the deficiency threshold). Application of potassium sulphate eliminated the potassium deficiency, but led to deficiency in Ca and Mg, and had no major effect on soil properties. The value of phonolite as a long term source of nutrition was questioned in terms of an observed release of sodium, but the rock used in this trial was unusually rich in sodium for a phonolite (4.37% Na; 5.9% Na<sub>2</sub>O); other phonolites, or related trachytes (which can be extremely K rich, up to 12-13% K<sub>2</sub>O; e.g. Sutherland, 1967) may not have this problem.

In a separate study involving pot trials with rice (*Oryza* sp.), Weerasuriya et al. (1993) used phlogopite mica (a magnesium equivalent to biotite, in which Fe is substuted entirely by Mg; Deer et al., 1992) and potassium feldspar, both treated with concentrated nitric acid to enhance the availability of K. Application rates were equivalent to 200 kg ha<sup>-1</sup> for phlogopite

and 500 kg ha<sup>-1</sup> for feldspar, corresponding to approximately 10 kg K ha<sup>-1</sup> as total K. Significant increases in rice yield (panicle number and seed weight) were observed for the use of phlogopite compared with both control and muriate of potash.

## 7.3 Trials with Feldspars

Other work with feldspars alone has been carried out in Egypt, in field trials with okra (Abdel-Mouty and El-Greadly, 2008), onions (Ali and Taalab, 2008) and tomatoes (Badr, 2006). In trials with okra (*Abelmoschus esculentus* cv Eskandrani and cv Balady), potassium feldspar was used as one of a number of treatments designed to assess crop response (Abdel-Mouty and El-Greadly, 2008). Significant increases in pod yield were observed in treatments with potassium feldspar compared with a control, and these increases were enhanced with foliar application of gibberellic acid. Similarly, Ali and Taalab (2008) report increasing individual bulb weight for onions (*Allium cepa* L.) with increasing feldspar application, and increasing overall yield, from 18 Tha<sup>-1</sup> with application equivalent to 95 kg K ha<sup>-1</sup>, to 30 T ha<sup>-1</sup> with 285 kg K ha<sup>-1</sup>. Crop yields were however lower than those recorded for equivalent chemical fertiliser applications, by about 15%.

In experiments with mixtures of compost and feldspars, Badr (2006) reports an increase in the amount of available K with increasing proportions of added feldspar. There was also a clear increase in tomato yield with increasing proportions of feldspar in the composts used, from yields of 27 t/ha (control) to 45 T ha<sup>-1</sup> with an application equivalent to 360 kg K ha<sup>-1</sup>. Potassium content of fruits also increased. Again, fruit yields were 10–15% less than observed for application of chemical fertiliser (in this case potassium sulphate).

The Egyptian experiments with potassium feldspars suggest that feldspar is almost as effective as a soluble potash fertiliser. Sanz Scovino and Rowell (1988) report trials with a mix of the grass *Brachiaria dyctioneura* and the legume *Pueraria phaseoloides* with the potassium feldspar sanidine, compared with muriate of potash. An increase in yield was observed, with maximum applications equivalent to 80 kg K ha<sup>-1</sup>, but this was statistically insignificant. Sanz Scovino and Rowell (1988) comment that application of potassium feldspar may be beneficial as an alternative to KCl in Colombia, where economic and agricultural conditions, including the occurrence of oxisol soils, give rise to disadvantages with the use of KCl.

## 7.4 Trials with Non-Granitic Bulk Rocks

Bulk rocks other than granites and rocks rich in alkali feldspars have been used as a source of soil nutrients in a number of studies, dominated by the use of phosphate rock which is well-established commercially (van Straaten, 2007). In general, trials with silicate rocks, especially volcanic rocks, have not focused on their possible value as a source of K, but as a source of a range of major and trace mineral nutrients.

Leonardos et al. (1987) approach the problem of nutrient supply in laterite systems, where soils (oxisols) are predominantly extremely depleted in nutrients, and have little or no capacity (by virtue of their mineralogy being dominated by quartz, kaolinite and aluminium oxy-hydroxides) to retain nutrients supplied by soluble fertilisers. Leonardos et al. (1987) strongly criticise conventional fertiliser practice that is driven by the industrial world: "Unfortunately, the standard concept and technology of soil fertilizer... is behind that of the superphosphate concept developed by J.B. Lawes in England, 150 years ago. ... Had this technology been originally developed for the deep leached laterite soils of the tropics instead for (*sic*) the glacial and rock-debris-rich soils of the northern hemisphere our present fertilizers might have been quite different."

Preliminary data for trials with beans (Phaseolus vulgaris), napier grass (elephant grass; Pennisetum purpureum), and Eucalyptus pellita were carried out near Brasilia, using a range of different rock types (Leonardos et al., 1987). For beans, the greatest yields were observed in experiments with basalt applied at the equivalent of 6 T ha<sup>-1</sup> and mica schist (as a source of K) applied at the equivalent of 60 kg K ha<sup>-1</sup>, with N and P applied as ammonium nitrate and triple superphosphate. For napier grass, yields increased with the application of limestone, basalt and the metamorphic rocks mica schist (mica-rich) and migmatite (feldsparrich), up to 30% greater than for a control. Eucalyptus' response was measured by comparing plant height and trunk diameter; both were greatest with application of basalt.

More recently, other experiments have been carried out using volcanic rock (ultramafic rocks from the Mata da Corda; Theodoro and Leonardos, 2006; 30-36% SiO<sub>2</sub>). This rock forms bedrock in an area of forest known for its fertility compared with neighbouring soils. Application at the rate of 2.5-3 T ha<sup>-1</sup> was recommended to farmers who conducted the trials, which were designed to raise awareness in local people of possible benefits of indigenous rock sources. Soil fertility improved, with measurable increases in Ca, Mg, P and K, and reduction in exchangeable Al. Results were presented based on records of application and yield supplied by participating farmers, including their observations of plant growth. Maize (Zea mays), sugar cane (Saccharum officinarum), rice (Oryzea sativa), manioc (Manihot esculenta) and corn (Zea mays) were used in the trials.

Theodoro and Leonardos (2006) found in every case that application of the rock fertiliser compared to conventional practice with chemical fertilisers gave increased crop yields, most notably for sugar cane. However, in some cases manure was used in combination with the rock dust, and so it is difficult to attribute the observed benefits to rock dust alone. Theodoro and Leonardos (2006) report that the reaction of farmers engaged in the project was very positive to rock dust, and that the project gave many valuable social benefits in addition to the perceived agronomic benefits. The use of rock dust in tropical soils is developed further (without evidence from trials) by Fyfe et al. (2006), on the basis of a range of societal benefits.

Other work has been carried out to determine the behaviour under soil conditions of basaltic rocks (45% SiO<sub>2</sub>; Hinsinger et al., 2001; Gillman et al., 2002) and andesitic rocks (typically 58% SiO<sub>2</sub>; Meheruna and Akagi, 2006).

In laboratory studies designed to understand weathering phenomena, Hinsinger et al. (2001) assessed the influence of plant growth on the weathering of basalt, and report the results of experiments to assess the extent to which crop plants enhance the release of cations from the rock. In these experiments, lupin (Lupinus albus cv Lublanc), oilseed rape (Brassica napus cv Drakkar), banana (Musa paradisiaca cv Cavendish CV901) and maize (Zea mays cv Mona) were compared. The plants were grown in contact with a mesh bag that contained the powdered basalt (0.1–0.2 mm, irrigated with a nutrient solution. The effluent solution was analysed for elements liberated from the basalt that were lacking in the irrigation solution (especially Fe, Ca, Mg, Na, Si). Plant uptake of these elements was monitored with time, so the rate of basalt weathering could be compared between species. Rates of release to the leaching solution were greatest for Mg and Na with lupins, and for Fe with maize. Mass balance calculations showed that maize had greatest uptake of Na, Fe and Si, and banana had greatest uptake of Mg, Fe and Ca. The calculated weathering rates for the basalt are compared in Fig. 6. Bearing in mind that the reaction rate is expressed as a logarithm, Figure 6 shows that weathering rate varies from one plant species to another, and differs for different elements. With the exception of Ca, all weathering rates increase in the presence of plant growth, by an order of magnitude for Fe, and by a factor of up to 5 for Mg and Na, for banana, lupin and maize.

Other laboratory experiments are reported by Gillman et al. (2002), who compared the effect of addition of basalt dust on the soil properties for a range



**Fig. 6** Comparison of dissolution rates for basalt in the presence of different crops (data from Hinsinger et al., 2001)

of Queensland soil types in experiments involving incubation for 3 months at field moisture capacity. Soil properties (pH, CEC, exchangeable cations, P sorption capacity, extractable P and extractable Si) were determined before and after leaching with the equivalent of 2750 mm rainfall. Exchangeable H and Al were reduced, with an increase in exchangeable Ca and Mg. Basalt released both Si and P to the soil, and gave increased extractable P.

Meheruna and Akagi (2006) carried out experiments similar to those of Hinsinger et al. (2001), using andesite instead of basalt, and growing rice (*Oryza sativa*), maize (*Zea mays*) and soybean (*Glycien max* L.) under hydroponic conditions. Weathering was determined from the release of Si, Ca, Mg, Mn, Fe and Al, which was enhanced in all cases, most notably for Ca and Mg with maize. Weathering rate data were not given, and so cannot be compared directly with results presented by Hinsinger et al. (2001).

Although crushed rock materials have been promoted as nutrient sources for some time, this has been largely through the alternative or organic farming sectors (e.g. Lisle, 1994; Walters, 1975). The use of rock dusts underpins the "Remineralize the Earth" movement (http://www.remineralize.org), but as yet there are no published, statistically sound, trials that clearly demonstrate the agronomic value of rock dusts in isolation from other applications to the land, such as manures or composts. The use of rock powders in farming is, however, sufficiently acceptable and widespread to sustain commercial activity, especially with the growth in popularity of organic food.
#### 8 Conclusion

Investigations of the possible use of silicate rocks as sources of K have yielded varying results. In general, there is a lack of consistency in terms of the design of individual trials, limiting comparison and extrapolation. Soil type and properties are rarely reported. Given its inherent inconsistency, the work reported here is limited in the extent to which it can be interpreted. However, it is undeniable that the price of potash has risen considerably since much of the experimental work was carried out, justifying further research.

Laboratory studies of the dissolution of feldspars and nepheline, with or without the presence of mycorrhizal soil microbial communities, have shown that in principle these minerals are capable of releasing K on a timescale that is appropriate to meet the needs of growing plants. When trials have been carried out, results have been variable, and have not always been substantiated by statistically rigorous experimental design or analysis. Nevertheless, the most positive results have been obtained for soils with a very low cation exchange capacity, typical of those of oxisols encountered in tropical/sub-tropical regions and that have poor nutrient retention characteristics. Thus the application of crushed rock materials, that might be derived from local sources, has to be considered seriously as an alternative to conventional chemical fertilisers where those might be poorly accessible on the grounds of cost or availability.

The price of conventional sources of K is much greater now than at the time of the trials reported here. Thus those reports that dismissed the use of crushed rock materials, especially granite, on economic grounds need to be reconsidered. In circumstances commonly encountered in developing countries, where the price of potash fertilisers on the world market is prohibitively high (especially following recent price rises), it is beneficial to explore the use of locally-sourced igneous rocks as a source of K.

There is clearly scope for careful selection of the silicate rock potassium source. On the basis of dissolution rate data, priority should be given to rocks that contain nepheline. These include nepheline syenites, phonolites and trachytes, rock types that are characteristic of rift valley tectonic settings, and so are available in many countries that might have difficulty funding adequate conventional fertiliser purchases. In geological descriptions, there is some overlap between trachyte and basalt, and areas mapped as basalt may include trachytic rock types. Thus the behaviour of basaltic rocks in crop production systems is relevant in consideration of sources of K, although such rocks generally have relatively low K contents.

If nepheline-bearing rocks are not available, it is appropriate to consider the use of potassium feldspars, one of the commonest minerals in the Earth's crust. Potassium feldspar often occurs in coarse grained pegmatites, from which it can easily be separated by eye, as well as in granites. Published trials in African and Columbian production systems demonstrate the benefit of application of feldspar for a range of crop types, as a material almost equivalent in performance to soluble potassium fertilisers.

For northern hemisphere countries in which agricultural soils are developed on tills or mechanically weathered rock substrates, or are rich in clay minerals, the ability to retain mineral nutrients is much greater than in tropical regions. In these circumstances, the use of crushed rocks as a source of K is less likely to be beneficial in general terms, although the long-term replenishment of K from crushed rock sources may be desirable in specific farming practices, such as organic farming.

One of the major problems associated with the use of crushed rocks as a source of K (or other nutrients) is the lack of robust experimental trials that are consistent in terms of both their agronomic and their mineralogical design. It is essential that the two disciplines of biology and mineralogy have equal weight in experiments of this type, as expertise in both is needed. Such trials require substantial investment in terms of time, and so need to be considered strategically. The current high cost of conventional potassium fertilisers coupled with increased demand for crop production both justify strategic investment of this type.

#### References

- Abdel-Mouty M.M., El-Greadly N.H.M. (2008) The productivity of two okra cultivars as affected by gibberilic acid, organic N, rock phosphate and feldspar applications, J. Appl. Sci. Res. 4, 627–636.
- Ali H.A., Taalab A.S. (2008) Effect of natural and/or chemical potassium fertilizers on growth, bulbs yield and some physical and chemical constituents of onion (*Allium cepa*, L.), Res. J. Agric. Biol. Sci. 4, 228–237.

- Badr M.A. (2006) Efficiency of K-feldspar combined with organic materials and silicate-dissolving bacteria on tomato yield, J. Appl. Sci. Res. 2, 1191–1198.
- Bakken A.K., Gautneb H., Myhr K. (1997) The potential of crushed rocks and mine tailings as slow-releasing K fertilizers assessed by intensive cropping with Italian ryegrass in different soil types, Nutr. Cycl. Agroecosys. 47, 41–48.
- Bakken A.K., Gautneb H., Sveistrup T., Myhr K. (2000) Crushed rocks and mine tailings applied as K fertilizers on grassland, Nutr. Cycl. Agroecosys. 56, 53–57.
- Barlow S.G. (1996) The mineralogical, geochemical and experimental evaluation of boulder clay from Adswood, Stockport, as a model brickclay raw material, PhD thesis (unpublished), University of Manchester, 313 p.
- Barral Silva M.T.B., Hermo B.S., Garcia-Rodeja E., Freire N.V. (2005) Reutilization of granite powder as an amendment and fertilizer for acid soils, Chemosphere 61, 993–1002.
- Blum A.E., Stillings L.L. (1995) Feldspar dissolution kinetics, Chemical weathering rates of silicate minerals, in: White A.F., Brantley S.L. (Eds.), Rev. Mineral. 31, 291–351.
- Blum W.E.H., Herbinger B., Mentler A., Ottner F., Pollak M., Unger, E., Wenzel W.W. (1989) The use of rock powders in agriculture. 1. Chemical and mineralogical compsition and suitability of rock powders for fertilization, Z. Pflanzenernah. Bodenk. 152, 421–425.
- Bolland M.D.A., Baker M.J. (2000) Powdered granite is not an effective fertilizer for clover and wheat in sandy soils from Western Australia, Nutr. Cycl. Agroecosys. 56, 59–68.
- Brantley S.L., Stillings L. (1996) Feldspar dissolution at 25 °C and low pH, Am. J. Sci. 296, 101–127.
- Coreonos C., Hinsinger P., Gilkes R.J. (1996) Granite powder as a source of potassium for plants: a glasshouse bioassay comparing two pasture species, Fertil. Res. 45, 143–152.
- Curtis C.D. (1976) Stability f minerals in surface weathering reactions: a general thermochemical approach, Earth Sci. Process. 1, 63–70.
- De Ruiter P.A.C. (1979) The Gabon and Congo basins salt deposits, Econ. Geol. 74, 419–431.
- Deer W.A., Howie R.A., Zussman J. (1992) An Introduction to the Rock Forming Minerals, 2nd ed., Longman, Harlow, 696 p.
- El Tabakh M., Utha-Aroon C., Schreiber B.C. (1999) Sedimentology of the Cretaceous Maha Sarakham evaporates in the Khorat Plateau of northeastern Thailand, Sediment. Geol. 123, 31–62.
- FAO (2008) Current world fertilizer trends and outlook to 2011– 12, Food and Agriculture Organization of the United Nations, Rome, 44 p.
- Fyfe W.S., Leonardos O.H., Theodoro S.H. (2006) Sustainable farming with native rocks: the transition without revolution, Anais da Academia Brasileira de Ciencias 78, 715–720.
- Gillman G.P., Burkett D.C., Coventry R.J. (2002) Amending highly weathered soils with finely ground basalt rock, Appl. Geochem. 17, 987–1001.
- Gleeson D.B., Clipson N., Melville K., Gadd G.M., McDermott F.P. (2005) Characterization of fungal community structure on a weathered pegmititic granite, Microb. Ecol. 50, 360–368.

- Goldich S.S. (1938) A study in rock weathering, J. Geol. 46, 17–58.
- Goldschmidt V.M. (1933) Grundlagen der quantitativen Geochemie, Fortschr. Mineral. Krist. Petrog. 17, 112.
- Harley A.D., Gilkes R.J. (2000) Factors influencing the release of plant nutrients from silicate rock powders: a geochemical overview, Nutr. Cycl. Agroecosys. 56, 11–36.
- Hinsinger P., Barros O.N.F., Beneditti M.F., Noack Y., Callot G. (2001) Plant-induced weathering of a basaltic rock: experimental evidence, Geochim. Cosmochim. Acta 65, 137–152.
- Hinsinger P., Bolland M.D.A., Gilkes R.J. (1996) Silicate rock powder: effect on selected chemical properties of a range of soils from Western Australia and on plant growth assessed in a glasshouse experiment, Fertil. Res. 45, 69–79.
- Lægrid M., Bøckman O.C., Kaarstad O. (1999) Agriculture, Fertilizers and the Environment, CABI Publishing, Wallingford, 294 p.
- Le Maitre R.W. (1976) The chemical variability of some common igneous rocks, J. Petrol. 17, 589–637.
- Lee M.R., Parsons I. (1995) Microtextural controls of weathering of perthitic alkali feldspars, Geochim. Cosmochim. Acta 59, 4465–4488.
- Leonardos O.H., Fyfe W.S., Kronberg B.I. (1987) The use of ground rocks in laterite systems – an improvement to the use of conventional soluble fertilizers, Chem. Geol. 60, 361–370.
- Lian B., Wang B., Pan M., Liu C., Teng H.H. (2008) Microbial release of potassium from K-bearing minerals by thermophilic fungus *Aspergillus fumigatus*, Geochim. Cosmochim. Acta 72, 87–98.
- Limão N. and Venables A.J. (2001) Infrastructure, geographical disadvantage, transport costs, and trade, The World Bank Economic Review 15, 451–479.
- Lisle H. (1994) The Enlivened Rock Powders. Acres USA, PO Box 8800, Metairie, Louisiana 70011, ISBN 0-911311-48-3, 194 p.
- Lloyd S.L. (1918) Mining and manufacture of fertilizing materials and their relation to soils, Van Nostrand, New York, 125 p.
- Longo R. and Sekkat K. (2004) Economic obstacles to expanding intra-African trade, World Development 32, 1309–1321.
- Manning D.A.C. (1995) Introduction to Industrial Minerals, Chapman and Hall, London, 276 p.
- Martin J.M., Meybeck M. (1979) Elemental mass-balance of material carried by major world rivers, Marien Chemistry 7, 173.
- Meheruna A. Akagi T. (2006) Role of fine roots in the plantinduced weathering of andesite for several plant species, Geochem. J. 40, 57–67.
- Mengel K., Rahmatullah (1994) Exploitation of potassium by various crop species from primary minerals in soils rich in micas, Biol. Fert. of Soils 17, 75–79.
- Mengel K., Rahmatullah, Doh H. (1998) Release of potassium from the sily and sand fraction of loess-derived soils, Soil Sci. 163, 805–813.
- Moores S. (2009a) Fertiliser loses its mineral appetite, Ind. Minerals, 36–41.
- Moores S. (2009b) Potash tunnel vision, Ind. Minerals, 56-62.
- Ober J.A. (2007) Potash, 2006 Minerals Yearbook, United States Geological Survey, Reston, VA.

- Riggs K.S., Syers J.K., Appleton J.D. (1993) Dissolution of potassium feldspar rocks in soils in open leaching and closed-incubation systems, J. Sci. Food Agr. 62, 409–412.
- Rittenhouse P.A. (1979) Potash and politics, Econ. Geol. 74, 353–357.
- Rosling A., Lindahl B.D., Taylor A.F.S., Finlay R.D. (2004) Mycelial growth and substrate acidification of ectomycorrhizal fungi in response to different minerals, FEMS Microbiol. Ecol. 47, 31–37.
- Rudnick R.L., Gao S. (2003) The Composition of the Continental Crust, pp. 1–64, in: Rudnick R.L. (Ed.), The Crust, Vol. 3, Treatise on Geochemistry, Holland H.D., Turekian K.K. (Eds.), Elsevier-Pergamon, Oxford.
- Sanz Scovino J.I., Rowell D.L. (1988) The use of feldspars as potassium fertilizers in the savannah of Columbia, Fertil. Res. 17, 71–83.
- Searls J.P. (2000) Mineral Industry Surveys, Potash, 1999 Annual Review, United States Geological Survey, Reston, VA.
- Sheldrick W.F., Lingard J. (2004) The use of nutrient audits to determine nutrient balances in Africa, Food Policy 29, 61–98.
- Sheldrick W.F., Syers J.K., Lingard J. (2002) A conceptual model for conducting nutrient audits at national, regional and global scales, Nutr. Cycl. Agroecosys. 62, 61–67.
- Sheldrick W.F., Syers J.K., Lingard J. (2003) Soil nutrient audits for China to estimate nutrient balances and output/input relationships, Agr. Ecosyst. Environ. 94, 341–354.
- Sposito G. (1989) The Chemistry of Soils, Oxford University Press, Oxford, 277 p.
- Sutherland D.S. (1967) A Note on the Occurrence of Potassium-Rich Trachytes in the Kaiserstuhl Carbonatite Complex, West Germany, Mineral. Mag. 36, 334–341.

- Theodoro S.H., Leonardos O.H. (2006) The use of rocks to improve family agriculture in Brazil, Anais da Academia Brasileira de Ciencias 78, 721–730.
- Tole M.P., Lasaga A.C., Pantano C., White W.B. (1986) The kinetics of dissolution of nepheline (NaAlSiO<sub>4</sub>), Geochim. Cosmochim. Acta 50, 379–392.
- van Straaten P. (2006) Farming with rocks and minerals: challenges and opportunities, Anais da Academia Brasileira de Ceincias 78, 731–747.
- van Straaten P. (2007) Agrogeology: the use of rocks for crops, Enviroquest Ltd., Cambridge, Ontario, 440 p.
- von Wilpert K., Lukes M. (2003) Ecochemical effects of phonolite rock powder, dolomite and potassium sulfate in a spruce stand on an acidified glacial loam, Nutr. Cycl. Agroecosys. 65, 115–127.
- Wallander H., Wickman T. (1999) Biotite and microcline as potassium surces in ectomycorrhizal and non-mycorrhizal *Pinus sylvestris* seedlings, Mycorrhiza 9, 25–32.
- Walters C. (editor) (1975) The Albrecht Papers, Volume 1 Foundation concepts, Acres USA, PO Box 8800, Metairie, Louisiana 70011, ISBN 0-911311-05-X, 515 p.
- Wang J.G., Zhang F.S., Cao Y.P., Zhang X.L. (2000) Effect of plant types on release of mineral potassium from gneiss, Nutr. Cycl. Agroecosys. 56, 37–44.
- Weerasuriya T.J., Pushpakumara S., Cooray P.I. (1993) Acidulated pegmatitic mica – a promising new multi-nutrient mineral fertilizer, Fertil. Res. 34, 67–77.
- White A.F. (1995) Chemical weathering rates of silicate minerals in soils, Chemical weathering rates of silicate minerals, in: White A.F., Brantley S.L. (Eds.), Rev. Mineral. 31, 407–461.
- Woolley A.R. (2001) Alkaline rocks and carbonatites of the world. Part 3: Africa, Geological Society of London, London, 372 p.

# **Glandless Seed and Glanded Plant Research in Cotton**

Yingfan Cai, Yongfang Xie, and Jinggao Liu

#### Contents

1	Introduction	206
2	The Relationship Between Gossypol and Glands: Diversity of Glands Occurence in <i>Gossypium</i> Species	206
3	Inheritance of Glands and Gossypol	207
4	Traditional Breeding of Glandless Seed/Glanded Cotton Plants	208
5	Terpenoid Aldehydes in Glands and the Biosynthetic Pathway of Gossypol and Associated Terpenoids	209
6	Molecular Cloning of Genes Associated with Gossypol and Glands	210
7	A Strategy for Eliminating Gossypol and Creating Glandless Seed-Glanded Cotton Plants	212
8	Conclusion	213
Re	eferences	214

Y. Cai (🖂)

Abstract Recently the world has been entangled by insufficient food such as the lack of rice which threatens the safety of world food and affect sustainable development of the world economy, resulting in rising of food price. To address this issue, cotton appears as a possible source of both fiber and food. The research in recent years indeed showed bright prospects for this expectation. However, gossypol stored in the glands of cotton is toxic to nonruminant animals and humans, which wastes large amounts of cottonseed protein that could potentially provide the annual protein requirements for half a billion people. Gossypium species are characterized by their lysigenous glands containing terpenoid aldehydes, important secondary phytoalexins consisting mainly of gossypol, which constitute one of the important plant's defense system against pests and diseases. The best approach to address this issue is to create glandless seed and glanded plant cotton. A breakthrough in this field would realise the fulfilment of making cotton both a fiber and a food crop, which would be a feat of great magnitude for sustainable development of agriculture. Research on the relationship between glands and their secondary inclusions at the molecular level would be one approach for genetic engineering to control the glands and gossypol content. In this article, we review recent progress on glands and gossypol content for diverse gland types in *Gossypium* species, inheritance of glands and gossypol content, traditional breeding of glandless seeds and glanded plant cotton, the terpenoid aldehyde biosynthesis pathway, molecular cloning of the related genes, the strategy for genetic engineering, and future prospects.

College of Bioinformation, Chongqing University of Posts and Telecommunications, Chongqing 400065, China e-mail: caiyf3000@yahoo.com.en

**Keywords** Cotton • Pigment gland • Gossypol • Inheritance and breeding • Gene cloning and genetic engineering

### **1** Introduction

Cotton, genus Gossypium L. containing 49 species, is a leading fiber and potential food crop, and cottonseed has compared favorably with other traditional food sources as a source of protein in several human nutritional studies (Lusas and Jividen, 1987). Cottonseed is among the most abundant protein meal and it represents 6.9% of world protein meal production only next to soybeans, rapeseed (Ash and Dohlaman, 2006). Global cottonseed production can potentially provide the annual protein requirements for half a billion people, for every kilogram of fiber produced by the plant, 1.65 kg of seed are also produced (Sunilkumar et al., 2006; Bertrand et al., 2005; Gerasimidis et al., 2007). But all cotton species have characteristic lysigenous glands containing terpenoid aldehydes, predominantly the sesquiterpenoid gossypol which is toxic to nonruminant animals and humans (Fryxell, 1968; Bell, 1969; Stipanovic et al., 1977). The pigment glands, as the storage organs of terpenoid aldehydes, including gossypol, are on the surfaces of cotton organs or tissues (Punit et al., 1991).

Gossypol was first characterized in a classic series of studies by Adams et al. in 1938 (Heinstein et al., 1962). Actually gossypol in cotton is a double-edged sword. It provides constitutive and inducible resistance against a variety of pests and diseases and acts as an important phytoalexin (Wang et al., 2004; Townsend et al., 2005). In vitro, gossypol also acts as an anticancer (Liu et al., 2002; Ye et al., 2007; Oliver et al., 2005), anti-HIV (Lin et al., 1993; Bourinbaiar Lee-Huang 1994; Keller et al., 2003), antibacterial and nonhormonal male contraceptive (Coutinho, 2002; Tegos et al., 2002; Lopez et al., 2005; Badawy et al., 2007). However, on account of its toxicity, the nutritional value of cottonseed is reduced much, in addition gossypol discolors cottonseed oil (Cherry, 1983) which results in a tremendous waste of seed protein and oil. To reduce gossypol in cottonseed, studies have developed glandless cotton, glandless both in seeds and plants, with no or very low gossypol content (McMichael, 1959; Vaissayre and Hau, 1985).

The protein and oil of the glandless seed are free of gossypol and can therefore be used directly as food (Lusas and Jividen, 1987), but plant resistance to pests is reduced and the yield of cotton fibers is reduced due to the absence of gossypol and other terpenoid aldehydes, the glandless cotton has been applied in very limited area (Cherry, 1983; Vaissayre and Hau, 1985).

Cotton has been anticipated to become a crop that produces both fiber and food for decades, and thus developing a cotton variety that has gossypol within glanded roots, leaves, and stems to maintain the resistance trait but has glandless seeds for safe food utilization is now necessary (Cai et al., 2004). In other words, the ideal cotton cultivar would delay gossypol biosynthesis until germination, a phenotype seen only in some Australian Gossypium species (Brubaker et al., 1996). Many studies have examined the relationship between cotton glands and gossypol production, and recent efforts have shown much progress in this field, especially in molecular biology and genetic engineering. Such breakthroughs will revolutionize cotton breeding, cotton planting, industrial processing of cotton seed, and even use as animal feed and human food, and eventually will be beneficial to sustainable development of agriculture and human being and guarantee the safety of the world food.

# 2 The Relationship Between Gossypol and Glands: Diversity of Glands Occurence in *Gossypium* Species

Gossypol is synthesized in cotton roots, and transported and stored within pigment glands of cotton above ground (Smith, 1962). The pigment glands are spherical on leaves, bracts, and ovary surfaces; elliptical on stems and stigma surfaces; and spherical to oval on calyx surfaces. Genotypes differ markedly in gland distribution (Punit et al., 1991). The content of gossypol in cotton is dependent on the genetic types of pigment glands. The gland number of cotyledonary leaves is significantly and positively associated with free gossypol and positively associated with the number of gossypol-producing glands and free gossypol content in seeds (Punit et al., 1995).

High performance liquid chromatography, HPLC, analysis of gossypol in different gland genotypes

of cotton revealed major differences among glands and species of Gossypium, including the glandless and glanded G. hirsutum L. and G. barbadense L. Especially G. hirsutum variety Xiangmian 18 is characterized by glandless seeds with very low gossypol content and low number of glands in foliage (Cai et al., 2004). "Glandless-seeded trait" in Australian G. sturtianum Willis reportedly lacks terpenoid aldehydes, and thus may represent an important genetic resource in the development of cottonseed oils and protein free of these toxins (Brubaker et al., 1996). The special trait of glands in this Australian species involves delayed pigment gland morphogenesis (Zhu et al., 1999). The studies on gossypol of the cotyledons during seed germination and the relationship between gossypol and gland formation in five wild Australia Gossypium species such as G. sturtianum, G. nandewarense Dereda, G. australe Mueller, G. nelsonii Fryxell, and G. bickii showed little gossypol in dormant seeds and cotyledons 24 h before seed germination. However, 48 h after germination, gossypol began to accumulate in cotyledons, although the content was very low. After 1-2 days of growth, the gossypol content in cotyledons increased markedly and was highest 9 days after germination. The dynamics of gossypol accumulation during seed germination was basically the same in all five wild species, but the time at which gossypol first appeared in the cotyledons differed significantly among them. Presumably some genetic regulation exists between glands and gossypol content (Zhu et al., 1999). Pigment glands in Gossypium species have diverse characteristics; glanded cotton normally contains gossypol in both seeds and other plant parts, and the content of gossypol is highly correlated with the number of pigment lands in glanded cotton (Singh and Weaver, 1972). However, in G. somalense Hutchinson (Genome E2) and G. herbaceum L. (Genome E1), pigment glands are found in the seeds but the gossypol content is unmeasurable.

In general, the following five types of pigment glands are found in *Gossypium* species: glandless cotton, with no or very low gossypol and no glands in seeds, roots, stems, and leaves; glanded cotton, with gossypol within glands in seeds, roots, stems, and leaves; cotton with a delayed gland morphogenesis trait (Zhu et al., 1999), specifically the Australian *Gossypium* species *G. bickii* (Genome G1), *G. australe* (Genome C3), and *G. sturtianum* (Genome C1-n), which contain no or very low gossypol and have no

glands in seeds, but have gossypol and glands in roots, stems, and leaves; cotton with glands in seeds, but very low gossypol content, e.g., *G. stocksii*; and cotton that has no or few glands and low gossypol in seeds, but has some glands and gossypol in plant parts. Hence, the diversity of glands in *Gossypium* and the relationship between glands and gossypol content warrant further exploration.

#### 3 Inheritance of Glands and Gossypol

Results on the genetics of glands in cotton have differed because different gland materials have been used. The glandless trait is thought to be controlled by two or three recessive or dominant genes in different gland types from different cotton species. According to McMichael (1960), the glandless trait is controlled by two recessive genes, gl2 and gl3, found in lines of Hopi cotton (McMichael, 1960). In other varieties, however, such as Hai 1 and others, it is controlled by dominant genes such as GL2e (Carvalho and de Vieira, 2000). Yet other scientists have reported that the glandless trait is a quantitative trait controlled by minor polygenes (Lee et al., 1968). GL2e in G. barbadense is dominant and produces glandless plants and seeds. The free and total gossypol contents in F1 seeds from this variety were similar to those of the glandless parent, showing complete dominance of this gene (Carvalho and de Vieira, 2000).

Twenty-one gossypol glanded cultivars from Turkey were crossed with two Turkish gossypol-free cultivars, Suz-86 and 129. Of 5588 plants raised in the F2 generation, 340 were entirely glandless. According to the two independent pairs of genes (glanded: GL2 GL2 GL3 GL3; glandless: gl2 gl2 gl3 gl3), the expected 1/16 ratio of glandless plants in the F2 generation was in accordance with a chi-square test (Arshad Mahmood et al., 2004).

A mutant of Xiang X9628 (Xiangmian 18) in upland cotton (*G. hirsutum*) has normal glanded leaves and a low gossypol content in seeds (Zhang et al., 2001). Genetic analysis indicated that the glanded leaves and low gossypol content of the mutant are controlled by two pairs of recessive duplicate genes. Allelic tests showed that one of these is allelic to the gl2 gene and the other is a multi-allele of the gl3 loci. This was therefore considered to be a new gland-forming gene in *Gossypium* and was named gln3. Data from crosses using line 247-1, which is rich in glands and has a high gossypol content in the flower buds, and plants homozygous for various combinations of gland-determining alleles at the gl1, gl2 and gl3 loci indicate that the high gossypol content of 247-1 could be explained by the presence of a high-potency allele at the gl3 locus, plus another factor or factors independent of the other gland-determining alleles (Wilson and Smith, 1976).

Studies on the inheritance of the high-glanding (HG) trait in cotton (*G. hirsutum*) using crosses made among HG, normal glanding (NG) and glandless genotypes, and with isolines of the HG breeding line XG15, which has the genotypes 2 (Gl2gl3) and 2 (gl2Gl3), showed that isoline XG15 gl2Gl3 expressed the HG phenotype, suggesting that HG is conferred by a special Gl3 allele derived from XG15. This allele, derived originally from Socorro Island cotton, was tentatively designated Gl3s (Calhoun, 1997). Additive effects accounted for more than 90% of the total genetic variance in seed gossypol level in all trials.

Epistatic effects, though small, were frequently significant. In *G. barbadense*, Gl2 and Gl3 were found to be associated with the production of similar amounts of gossypol, whereas previous trials with cultivated varieties of *G. hirsutum* showed that more than twice in Gl2 was expressed compared to Gl3. The higher average productivity of seed gossypol in cultivated *G. barbadense* compared to *G. hirsutum* is attributable to higher activity at the Gl3 locus in the former species (Lee, 1973).

Additive and dominance effects and other digenic interactions are associated with gossypol content, with the additive effects and additive  $\times$  additive epistasis being more prominent. The variety of interactions indicate that the major gland determining the loci gl2 and gl3 are not the only determinants of gossypol content, but that the trait shows polygenic inheritance (Singh et al., 1991).

### 4 Traditional Breeding of Glandless Seed/Glanded Cotton Plants

Since a close relationship exists between gossypol content and pigment glands, tactics to control gossypol through selecting and controlling the gland trait have proven to be appropriate and effective. In cotton breeding, the gland trait has been used for two different purposes: to produce glands rich in gossypol or with high glanding (HG) or to produce glandless plants.

On the one hand, breeding and application of a variety with a high density of glands and high gossypol content can enhance cotton plant resistance to pests and diseases. Such lines and varieties with high gossypol are beneficial as they reduce the cost for pest control and therefore cause less environmental pollution from the use of chemicals (Calhoun and Jones, 1994; Calhoun, 1997; Lee, 1978; McCarty et al., 1996; Bourland and Benson, 2002; Li et al., 1996).

On the other hand, to make full use of cottonseed protein and oil, scientists bred the glandless cotton 23B, glandless both in seeds and plants, with no or very low gossypol content (McMichael, 1954, 1959, 1960). Subsequently, many more glandless varieties have been bred from the resources created by McMichael (Lusas and Jividen, 1987; Miravalle, 1972; Vaissayre and Hau, 1985). Cottonseed compares favourably to other traditional food sources as a source of protein in several human nutrition studies, and glandless cottonseed is safe and nutritious (Lusas and Jividen, 1987). However, glandless varieties have proven to be commercially unviable as they are more susceptibility to insect pests because of the systemic absence of glands that contain gossypol and other protective terpenoids (Sunilkumar et al., 2006). Hence, the best choice currently available is plants with glandless seeds and glanded foliage, and many scientists have attempted to breed such plants.

A new upland cotton (*G. hirsutum*) cultivar has been bred in China, Xiangmian 18, with glandless seed, which has very low seed gossypol and is a glanded plant. This was developed from Xiangmian  $10 \times \text{Zhong 5655}$ , and was released in China during 2000. The glandless traits of glandless Zhong 5655 originated from Hai 1 (*G. barbadense* L.) This new cultivar can be grown for the production of seed protein and oil as well as lint (Zhang X. et al., 2001).

To transfer the traits for glandless seeds and glanded plants – so-called "delayed pigment gland morphogenesis" or "delayed development of gossypol glands" – Australian species have been bred to cultivated species of *Gossypium* to produce tri-species hybrids. A series of hybrid seeds or materials with the traits for glandless seeds and glanded plants have been obtained, but these have not yet reached the potential

for use in commercial cotton production because of many problems such as sterility, with wild traits and other disadvantages (Altman et al., 1987; Kulkarni et al., 2002; Mergeai, 1992; Mergeai et al., 1995, 1997; Dilday, 1986; McCarty et al., 1996; Vroh et al., 1999; Bi et al., 1998, 1999; Zhu et al., 1993, 2004, 2005; Bourland and Jones, 2006; Ahoton et al., 2003).

# 5 Terpenoid Aldehydes in Glands and the Biosynthetic Pathway of Gossypol and Associated Terpenoids

Lysigenous glands in cotton plants contain terpenoid aldehydes (Bell, 1969; Stipanovic et al., 1977). In cultivated cotton, glands in achlorophyllous plant parts of G. hirsutum predominantly contain the gossypol and its methyl and dimethyl ethers are found in G. barbadense. Glands in young green tissues of G. hirsutum, however, contain hemigossypolone as the predominant terpenoid aldehyde, and a new quinone, hemigossypolone-7-methyl ether occurs in G. barbadense (Bell et al., 1978; Benedict et al., 2004). A survey of terpenoid quinones and their heliocide derivates in wild Gossypium spp. and related genera in the Gossypieae showed considerable diversity. Several cadinane sesquiterpenoids and heliocides (sesterterpenoids) are deposited in the pigment glands of cotton plants, which function in pathogen and insect resistance (Stipanovic et al., 1999; Martin et al., 2003).

Based on the structure of gossypol, it was suggested that it might be formed metabolically from acetate via the isoprenoid pathway. Heinstein and colleagues examined the location of incorporated <sup>14</sup>C and studied the incorporation of mevalonate-2-<sup>14</sup>C, a key intermediate in the isoprenoid pathway, into gossypol, as well as the distribution of radioactivity in the gossypol molecule (Heinstein et al., 1962, 1970).

The cotton terpenoid aldehydes and cadalene derivatives are sesquiterpenes (C15) derived from a cytosolic branch of terpenoid metabolism via the mevalonate pathway (Heinstein et al., 1962, 1970). Farnesyl diphosphate (FPP) is generated as the linear carbon skeleton of the sesquiterpenes in cotton (Essenberg et al., 1985; Stipanovic et al., 1986; Liu et al., 1999). The various sesquiterpene synthases

cyclize FPP to form the molecular frameworks of different sesquiterpene types (Gershenzon and Croteau, 1993).

Gossypol is the sesquiterpenoid formed in cottonseed, with only traces of desoxyhemigossypol (dHG) and hemigossypol. In cotton foliage, hemigossypolone is formed from dHG (Stipanovic et al., 1999). The cadinene enzyme was first purified from a glandless cotton mutant by Davis and Essenberg (1995) as a soluble hydrophobic monomer with a molecular mass of 64 to 65 kD.

As reported by Martin et al. (2003), there are several cadinane sesquiterpenoids and heliocides (sesterterpenoids) deposited in pigment glands in cotton plants that function in pathogen and insect resistance (Stipanovic et al., 1999). A proposed pathway for the biosynthesis of these compounds is shown in Fig. 1. Infection of cotton stele tissue with conidia of Verticillium dahliae that induced the formation of sesquiterpenoid phytoalexins also induced 3-hydroxy-3-methylglutaryl-CoA reducterpenoid aldehydese (HMGR) mRNA and HMGR activity, demonstrating the important role of HMGR in the biosynthesis of the sesquiterpenoids (Bianchini et al., 1994). Studies with specifically labeled mevalonic acid (MVA) or acetate demonstrated the folding pattern of farnesyl diphosphate (FDP) required for gossypol formation. Subsequently, the enzymatic product of the cyclization of E, E-FDP in cotton extracts was identified as (+)- $\delta$ -cadinene (CDN) (Benedict et al., 2001; Davis and Essenberg, 1995; Chen Z.Y. et al., 1995). The enzymatic mechanism of CDN synthase for the formation of the cadinane structure of cotton sesquiterpenoids was shown to involve the isomerization of FDP to a nerolidyl intermediate, cyclization to a cis-germacradienyl cation, a 1, 3-hydride shift, cyclization to a cadinanyl cation and deprotonation to form (+)- $\delta$ -cadinene (Benedict et al., 2001). CDN synthase catalyses the committed step in the formation of the cadinane sesquiterpenoids from FDP at a branch point in the MVA pathway. Gossypol is the sesquiterpenoid formed in cottonseed with only traces of desoxyhemigossypol (dHG) and hemigossypol. In cotton foliage hemigossypolone is formed from dHG (Stipanovic et al., 1999). A Diels-Alder reaction accounts for the cycloaddition of myrcene or  $\beta$ -ocimene to hemigossypolone to form heliocides (Stipanovic, 1992). The monoterpenes or its precursors are synthesized by the 1-deoxy--xylulose-5-phosphate (DOXP)



**Fig. 1** The biosynthesis of gossypol in cottonseed and gossypol, hemigossypolene and heliocides<sub>1-4</sub> in cotton leaves (Modified from Martin et al., 2003)

pathway in the plastids (Martin et al., 2003; Essenberg et al., 1985; Bell, 1986; Stipanovic et al., 1986; Benedict et al., 1995; Davis and Essenberg, 1995; Alchanati et al., 1998).

In recent years, the biosynthetic pathway of gossypol and its derivates has been further elucidated. 8-Hydroxy-(+) d-cadinene is a precursor of hemigossypol in G. hirsutum (Yan-Hong wang et al., 2003). A (+) d-cadinene-8-hydroxylase, a cytochrome P450 mono-oxygenase of cotton sesquiterpene biosynthesis (CYP706B1), was expressed in aerial tissues of glanded cotton cultivars, but not or at an extremely low level in the aerial tissues of a glandless cultivar. The expression pattern of CYP706B1 and the position at which it hydroxylates (+) d-cadinene suggest that it catalyzes an early step in gossypol biosynthesis, and thus CYP706B1 holds good potential for manipulating gossypol levels in cottonseed via genetic engineering (Luo et al., 2001). Desoxyhemigossypol is a key intermediate in the biosynthesis of these compounds. A methyltransferase (S-adenosyl-L-Met: desoxyhemigossypol-6-O-methyltransferase) was isolated, purified, and characterized from cotton stele tissue infected with *V. dahliae*. Desoxyhemigossypol-6-methyl ether leads to the biosynthesis of methylated hemigossypol, gossypol, hemigossypolone, or the heliocides (Liu et al., 1999, 2005).

### 6 Molecular Cloning of Genes Associated with Gossypol and Glands

Based on elucidation of the terpenoid biosynthetic pathway, the related genes have also been cloned. The cadinene gene was first cloned and functionally characterized from the A-genome diploid cotton *G. arboreum* and comprise a large multigene family in cotton (Chen Z.Y. et al., 1995; Davis et al., 1996; Meng et al., 1999; Tan et al., 2000) similar to the terpene cyclase genes found in other plants (Facchini and Chappell, 1992; Back and Chappell, 1995). Several allelic and gene family variants of the cotton cadinen genes have since been isolated from both *G. arboreum* (Chen et al.,

Subfamily or gene name	Accession number	Gossypium species	Author	Submited year
cdn XC14	U23205	G. arboreum	Chen Z.Y. et al.	1995
cdn XC1	U23206	G. arboreum	Chen Z.Y. et al.	1995
cad1-b	X95323	G. arboreum	Chen X.Y. et al.	1996
cdn A	U27535	G. arboreum	Chen Z.Y. et al.	1995
cad1-A	Y18484	G. arboreum	Liang	1998
cad1-A	X96429	G. arboreum	Chen X.Y. et al.	1996
cad1-A	AF456410	G. barbadense	Cai	2002
cad XC14	AF453326	G. hirsutum	Cai	2001
cad1-C2	Y16432	G. arboreum	Meng et al.	1999
cad1-C1	AF174294	G. arboreum	Tan et al.	2000
cdn1	U88318	G. hirsutum	Davis	1997
cdn1-C4	AF270425	G.hirsutum	Townsend	2000
cdn1-C5	AY800106	G. hirsutum	Townsend	2004
cdn1-D1	AY800107	G. hirsutum	Townsend	2004

**Table 1** The cadinene genes and complementary DNA cloned in *Gossypium* in GenBank

1996; Liu et al., 1999; Meng et al., 1999; Tan et al., 2000) and the allotetraploid (A + D genomes) species *G. hirsutum* (Davis et al., 1998) (Table 1).

The cadinene enzyme and transcripts are induced in cotton stems infected with V. dahliae (Benedict et al., 1995; Alchanati et al., 1998; Bianchini et al., 1999; Tan et al., 2000), cotton suspension cultures treated with V. dahliae elicitors (Chen Z.Y. et al., 1995; Chen X.Y. et al., 1996; Liu et al., 1999), and cotton cotyledons infected with Xcm (Davis and Essenberg, 1995; Davis et al., 1996). cadinene is also developmentally regulated and cadinen transcripts increase during seed development in association with the biosynthesis and deposition of gossypol in the lysigenous storage glands of the embryo (Meng et al., 1999; Martin et al., 2003). Two major subfamilies of the Gossypium cadinene multigene family, cdn1-A and cdn1-C, have been proposed based on sequence relatedness, and appear to be differentially regulated at the transcriptional level (Meng et al., 1999; Tan et al., 2000). A third subfamily, cdn1-B, is represented by a single genomic clone.

A phylogenetic tree of the cotton cadinene genes showed the relatedness of members within the cadinene subfamilies and the distance between the different subfamilies (Fig. 2). cdn1-D1 is sufficiently different from the other cadinene genes of cotton to suggest that it belongs to another subfamily, designated cdn1-D.

The genetic structure of cadinene genes is well conserved, including the number, positions, and sizes of exons and introns, and also corresponds well to the genomic clones of other terpene cyclase genes, such as tobacco (*Nicotiana tabacum* L.) 5-epi-aristolochene synthase. Approximately 2 kb of promoter regions from each cadinene genomic clone were sequenced and a very low level of sequence conservation was found between all cadinene promoter regions. We found discrete regions of similarity, mostly in the vicinity of the predicted TATA box and transcription start site, but the similarity did not extend appreciably upstream (Back and Chappell, 1995).

Luo et al. (2001) cloned and identified the function of a gene of (+) delta-cadinene-8-hydroxylase, a cytochrome P450 mono-oxygenase of cotton sesquiterpene biosynthesis (Luo et al., 2001). Liu et al. (2005) identified and cloned a 1.9 -kb P450 coding for a 522 amino acid protein that is 48% identical to a soybean cytochrome P450 82 A3 and contains a conserved heme-binding motif and a consensus oxygen-binding pocket sequence (Liu et al., 2005). This P450 is expressed in the leaves of the glanded cotton *G. hirsutum*, but not in the leaves of glandless cotton. The leaves of glandless plants are devoid of glands and the associated terpenoids. This suggests that this P450 enzyme is involved in the terpenoid biosynthetic pathway in cotton.

Studies on gland-morphogenesis-related genes, applying suppressive subtractive hybridization and other methods, have successfully constructed a subtractive library and a normalized cDNA library from a cotton mutant, Xiangmian 18. Some relevant genes, such as the RanBP2 zinc finger protein gene in upland cotton, have been cloned (Cai et al., 2003; Chang et al., 2007; Xie et al., 2007). In addition, the gene of desoxyhemigossypol-6-O-methyltransferase



Fig. 2 Phylogenetic tree of cadinene protein sequences showing the subfamilies cdn1-A, cdn1-B, cdn1-C, and cdn1-D

from *G. barbadense* (Liu et al., 1999) and gene of cytochrome P450 related to glands have also been cloned and studied (Liu et al., 2005, 2007).

# 7 A Strategy for Eliminating Gossypol and Creating Glandless Seed-Glanded Cotton Plants

Traditional breeding is still the main approach in the breeding of cotton with glandless seed and glanded plant, but it is limited by, for example, long breeding cycles, frequent attachment to inappropriate genes, and especially, crosses between species. In practice, progress in traditional breeding to create glandless seed and glanded foliage cotton is very slow and large problems still exist. Genetic engineering can hopefully solve that. Advances in understanding the terpenoid biosynthetic pathway and cloning of the relevant genes provide tools to manipulate the biosynthesis of these defense terpenoids through genetic engineering creating glandless seed and glanded foliage cotton. Three aspects are involved in the strategy to eliminate the toxic cottonseed gossypol through genetic engineering: silencing gossypol-related genes in the seeds of glanded cotton using antisense and RNAi method; turning on related foliage genes in glandless cotton; and creating cotton only containing (+) gossypol, the toxicity of which is very low compared to (-) gossypol.

Antisense expression of cadinene genes was envisaged as a way to activate this silencing mechanism in cotton by blocking the cadinane-type sesquiterpene pathway and abolishing gossypol production in the transformants. An antisense construct of cdn1-Cl was introduced into cotton plants (Martin et al., 2003), and Southern analyses showed integration of antisense cdn1-C1 cDNA driven by the CaMV 35S promoter into the cotton genome. Northern blots demonstrated the appearance of cdn synthase mRNA preceding CDN synthase activity and the formation of gossypol in developing cottonseed. T2 cottonseed had reduced CDN synthase activity and up to a 70% reduction in gossypol. In T1 leaves, the accumulated gossypol, hemigossypolone, and heliocides were reduced by 92.4, 83.3, and 68.4%, respectively. These results demonstrate that the integration of antisense cdn1-C1 cDNA into the cotton genome leads to a reduction in CDN synthase activity and negatively impacts the biosynthesis of cadinane sesquiterpenoids and heliocides in cotton plants.

Another new cotton variant from the progeny of hemizygous G. hirsutum cv. Coker 312 transformed with antisense (+)- delta-cadinene synthase cDNA has reduced the TA levels (Benedict et al., 2004). The gossypol content in seeds of the variant was markedly lower than in seeds of T1 antisense plants. Eightynine percent of the variant seed had a 71.1% reduction in gossypol, and foliage of the variant plants had a 70% reduction in gossypol and a 31% reduction in heliocides. Compared to non-transformed plants, no reduction occurred in the number of lysigenous glands in the seed of the variant. The cotton variant showed uncoupling of TA synthesis and gland formation. In this case, the study assumed that the cotton variant may have resulted from somaclonal variation occurring in callus tissue during the transformation-regeneration process.

Constructs for the constitutive or seed-specific antisense suppression of cdn1-C4 were introduced into cotton by Agrobacterium-mediated transformation (Townsend et al., 2005). Gossypol levels were not reduced in the seeds of transformants with either construct, and the induction of cadinene expression did not affect stems of the constitutive antisense plants infected with V. dahliae. However, the induction of cadinene mRNA and protein in response to bacterial blight infection of cotyledons was completely blocked in the constitutive antisense plants. These results suggest that cdn1-C4 may be involved specifically in the bacterial blight response. Moreover, the cadinene multigene family may comprise a complex set of genes differing in their temporal and spatial regulation, and they may be responsible for different branches of the cotton sesquiterpene pathway. The antisense approach, however, has been limited or somewhat ambiguous (Martin et al., 2003; Townsend et al., 2005; Sunilkumar et al., 2006).

Recently, a study successfully used RNAi and a seed-specific promoter from the cotton a-globulin B gene to disrupt gossypol biosynthesis in cottonseed by interfering with expression of the cadinene synthase gene during seed development (Sunilkumar et al., 2006). The study demonstrated the possibility of significantly lowering cottonseed- gossypol levels in a stable and heritable manner. Results from enzyme activity and molecular analyses on developing transgenic embryos were consistent with the observed phenotype in mature seeds. Most importantly, the levels of gossypol and related terpenoids in foliage and floral parts were not diminished, and thus their potential function in plant defense against pests and diseases remained untouched. This research provides evidence for spatial and temporal confinement of RNAi-mediated suppression of the s-cadinene synthase gene in cottonseeds that contain the transgene. The results demonstrate the feasibility of a targeted RNAi-based approach to solve an age-old problem of cottonseed toxicity and provide an avenue to exploit the considerable quantities of protein and oil available in the global cottonseed output. The GL genes will be cloned and applied in genetic engineering to obtain glandless seed and glanded foliage cotton.

Since the (+) enantiomer shows little if any toxicity to nonruminant animals, cottonseed with high -protein content is underutilized due to toxic (-)- gossypol. The data from extracts of the Marie Galante cotton variety and the fact that the intact seed contain 95% (+) gossypol suggest a regio-stereoselective bimolecular coupling of hemigossypol to gossypol (Benedict et al., 2006). This can be used in future genetic engineering to eliminate the toxic (-) gossypol. As the (-) enantiomer has potential medicinal uses, cotton-seed with >95% (-) gossypol could have biopharmaceutical applications and can also be created through genetic engineering.

#### 8 Conclusion

With the increase in world population, decrease in available arable land, and need for more energy from plants, the world has been threatened by insufficient food. The desire to make cotton both a fiber and a food crop becomes more urgent. Peaple have bended themselve to this research for decades and have attempted many approaches to get rid of toxic gossypol and have made much progress as described above. In recent years, one successful research trend on glands and gossypol in cotton has focused on molecular cloning and genetic engineering to produce glandless seed and glanded foliage to abolish toxic seed gossypol beyond the immediate application of eliminating gossypol from cotton seeds, one can readily anticipate other application for the gene and its promoter. For example, the promoter could be used to introduce pesticides such as the BT toxin exclusively into the foliage (Liu et al., 2007). One could even use an exotic toxin, for example, from scorpions, since the green tissue special promoter would be directly expressed exclusively in the foliage. Obtaining such a promoter also provides a tool to manipulate the biosynthesis of HQG and heliocides through genetic engineering (Liu et al., 2007). The long-term goal is to elucidate the gene network mechanism controlling glands and gossypol while increasing the resistance of cotton to pests and pathogens, thus expanding utilization and the commercial value of cottonseed. Glandless seed and glanded foliage cotton will bring a revolutionary change in cotton breeding, cotton planting, and industrial processing of cotton seeds, even for use as a food supply. Cotton will become a crop that produces fiber, food, and oil simultaneously, which will help solve the problem of world food supply shortages.

At the same time, the research has also helped us to obtain a better understanding of the relationship between secondary compounds and storage organs (tissues) in plants and the molecular mechanisms controlling them. For example, artemesin and glands in southernwood (an antimalarial) have a similar relationship to gossypol and glands in cotton (Xu et al., 2003). The long-term goal of our research is to clarify the molecular mechanism (s) of the genetic control networks involved in secondary substances and storage organs in plants, and ultimately to control the biosynthesis of useful secondary compounds that will promote the sustainable development of agriculture and benefit humans.

Acknowledgements This work was supported by grants from the National Natural Science Foundation of China (No. 30771311 and No. 30440032) and Natural Sciences Foundation of Chongqing cstc 2007 BB1328 of China and USDA and Congress of USA and Nan'an committee of Science and Technology of 2008 in Chongqing.

#### References

- Ahoton L., Lacape J.M., Baudoin J.P., Mergeai G. (2003) Introduction of Australian Diploid Cotton Genetic Variation into Upland Cotton, Crop Sci. 43, 1999–2005.
- Alchanati I., Patel J.A.A., Liu JingGao, Benedict C.R., Stipanovic R.D., Bell A.A., Cui YunXing, Magill C.W. (1998) The enzymatic cyclization of nerolidyl diphosphate by delta-cadinene synthase from cotton stele tissue infected with Verticillium dahliae, Phytochemistry 47, 961–967.
- Altman D.W., Stelly D.M., Kohel R.J. (1987) Introgression of the glanded-plant and glandless-seed trait from *Gossypium sturtianum Willis* into cultivated Upland cotton using ovule culture, Crop Sci. 27, 880–884.
- Arshad Mahmood, Emiroglu S.H., Amjad A., Umer F. (2004) Genetic studies of gossypol character in some cotton varieties of Turkey, J. Anim. Plant Sci. 14, 18–19.
- Ash M., Dohlman E. (2006) Oil crops situation and outlook yearbook, Electronic outlook report from the Economic Research Service, United States Department of Agriculture.
- Back K., Chappell J. (1995) Cloning and bacterial expression of a sesquiterpene cyclase from Hyoscyamus muticus and its molecular comparison to related terpene cyclases, J. Biol. Chem. 270, 7375–7381.
- Badawy S.Z., Souid A.K., Cuenca V., Montalto N., Shue F. (2007) Gossypol inhibits proliferation of endometrioma cells in culture, Asian J. Androl. 9, 388–393.
- Bell A.A. (1969) Phytoalexin production and Verticillium wilt resistance in cotton, Phytopathology 59, 1119–1127.
- Bell A.A. (1986) Physiology of secondary products, in: Mauney J.R., Stewart J.M. (Eds.), Cotton Physiology, Cotton Foundation, Memphis, TN, pp. 597–621.

- Bell A.A., Stipanovic R.D., O'Brien D.H., Fryxell P.A. (1978) Sesquiterpenoid aldehyde quinones and derivatives in pigment glands of *Gossypium*, Phytochemistry 17, 1297–1305.
- Benedict C.R., Alchanati I., Harvey P.J., Liu J., Stipanovic R.D., Bell A.A. (1995) The enzymatic formation of delta-cadinene from farnesyl diphosphate in extracts of cotton, Phytochemistry 39, 327–331.
- Benedict C.R., Liu J.G., Stipanovic R.D. (2006) The peroxidative coupling of hemigossypol to (+)- and (-)-gossypol in cottonseed extracts, Phytochemistry 67, 356–361.
- Benedict C.R., Lu J.-L., Pettigrew D.W., Liu J., Stipanovic R.D., Williams H.J. (2001) The cyclization of farnesyl diphosphate and nerolidyl diphosphate by a purified recombinant d-cadinene synthase, Plant Physiol. 125, 1754–1765.
- Benedict C.R., Martin G.S., Liu J.G., Puckhaber L., Magill C.W. (2004) Terpenoid aldehyde formation and lysigenous gland storage sites in cotton: variant with mature glands but suppressed levels of terpenoid aldehydes, Phytochemistry 65, 1351–1359.
- Bertrand J.A., Sudduth T.Q., Condon A., Jenkins T.C., Calhoun M.C. (2005) Nutrient content of whole cottonseed, J. Dairy Sci. 88, 1470–1477.
- Bi I.V., Baudoin J.P., Mergeai G. (1998) Cytogenetics of the 'glandless-seed and glanded-plant' trait from Gossypium sturtianum Willis introgressed into upland cotton (Gossypium hirsutum L.), Plant Breeding 117, 235–241.
- Bi I.V., Baudoin J.P., Hau B., Mergeai G. (1999) Development of high-gossypol cotton plants with low-gossypol seeds using trispecies bridge crosses and in vitro culture of seed embryos, Euphytica 106, 243–251.
- Bianchini G.M., Benedict C.R., Joost O., Magill C. (1994) Induction of HMGR by Verticillium dahliae, in: Jividen, G.M., Benedict C.R. (Eds.), Biochemistry of Cotton, Cotton Incorporated Press, Raleigh, NC, pp. 41–46.
- Bianchini G.M., Stipanovic R.D., Bell A.A. (1999) Induction of delta-cadinene synthase and sesquiterpenoid phytoalexins in cotton by Verticillium dahliae, J. Agr. Food Chem. 47, 403–4406.
- Bourinbaiar A.S., Lee-Huang S. (1994) Comparative in vitro study of contraceptive agents with anti-HIV activity: gramicidin, nonoxynol-9, and gossypol, Contraception 49, 131–137.
- Bourland F.M. Benson N.R. (2002) Registration of Arkot 8727, a High Glanding Cotton Germplasm Line, Crop Sci. 42, 1384–1388.
- Bourland F.M., Jones D.C. (2006) Registration of Arkot 9406ne, Arkot 9605ne, and Arkot 9631ne, three nectariless germplasm lines of cotton, Crop Sci. 46, 1833–1834.
- Brubaker C.L., Benson C.G., Miller C., Leach D.N. (1996) Occurrence of terpenoid aldehydes and lysigenous cavities in the 'glandless' seeds of Australian *Gossypium* species, Aust. J. Bot. 44, 601–612.
- Calhoun D.S. (1997) Inheritance of high glanding, an insect resistance trait in cotton, Crop Sci. 37, 1181–1186.
- Calhoun D.S., Jones J.E. (1994) Proceedings Beltwide Cotton Conferences, January 5–8, San Diego, CA, USA. National Cotton Council, Memphis, USA, pp. 655–657.
- Cai Y., Mo J., Zeng Y., Ren W., Xu Y., Wang S., Chen F. (2003) Cloning of cDNAs of Differentially Expressed Genes in the Development of special Pigment Gland of Cotton by

Suppression Subtractive Hybridization, J. Beijing University Forest. 25, 6–10.

- Cai Y., Zhang H., Zeng H., Mo J., Bao J., Miao C., Bai J., Yan F., Chen F. (2004) An optimized gossypol highperformance liquid chromatography assay and its application in evaluation of different gland genotypes of cotton, J. Biosciences 29, 67–71.
- Carvalho L.P., de Vieira R. (2000) Expression of the Gossypium barbadense GL2E gene in Gossypium hirsutum annual cotton, Rev. Oleaginosas Fibrosas 4, 39–44.
- Chang P.-A., Li B., Xie Y.-F., Ni X.-M., Cai Y.F. (2007) Molecular cloning and expression analysis of a RanBP2 zinc finger protein gene in upland cotton (*Gossypium hirsutum* L.), Colloid. Surface. B 55, 153–158.
- Chen X.-Y., Wang M., Chen Y., Davisson V.J., Heinstein P. (1996) Cloning and heterologous expression of a second (+)-delta-cadinene synthase from *Gossypium arboretum.*, J. Nat. Prod. 59, 944–951.
- Chen Z.Y., Chen Y., Heinstein P., Davisson V.J. (1995) Cloning, expression, and characterization of (+)- delta -cadinene synthase: a catalyst for cotton phytoalexin biosynthesis, Arch. Biochem. Biophys. 324, 255–266.
- Cherry J.P. (1983) Cottonseed oil, J. Am. Oil Chem. Soc. 60, 360–367.
- Coutinho E.M. (2002) Gossypol: a contraceptive for men, Contraception 65, 259–263.
- Davis E.M., Tsuji J., Davis G.D., Pierce M.L., Essenberg M. (1996) Purification of (+)-δ-cadinene synthase, a sesquiterpene cyclase from bacteria-inoculated cotton foliar tissue, Phytochemistry 41, 1047–1055.
- Davis E.M., Chen Y.-S., Essenberg M., Pierce M.L. (1998) cDNA sequence of a (+)-delta-cadinene synthase gene (accession No. U88318) induced in *Gossypium hirsutum* L. by bacterial infection, Plant Physiol. 116, 1192.
- Davis G.D., Essenberg M. (1995) (+)- $\delta$ -Cadinene is a product of sesquiterpene cyclase activity in cotton, Phytochemistry 39, 553.
- Dilday R.H. (1986) Development of a cotton plant with glandless seeds, and glanded foliage and fruiting forms, Crop Sci. 26, 639–641.
- Essenberg D., Essenberg M. (1995) (+)-δ-Cadinene is a product of sesquiterpene cyclase activity in cotton, Phytochemistry 39, 553–558.
- Essenberg M., Stoessl A., Stothers J.B. (1985) The biosynthesis of 2, 7-dihydroxycadalene in infected cotton cotyledons: the folding pattern of the farnesol precursor and possible implications for gossypol biosynthesis, J. Chem. Soc. Chem. Comm. 9, 556–557.
- Facchini P.J., Chappell J. (1992) Gene family for an elicitor induced sesquiterpene cyclase in tobacco, Proc. Natl Acad. Sci. (USA) 89, 11088–11092.
- Fryxell P.A. (1968) A redefinition of the tribe Gossypieae, Bot. Gaz. 129, 296–308.
- Gerasimidis K., Fillou D.T., Babatzimcpoulou M., Ssou K., Katsikas H. (2007) Preparation of an edible cottonseed protein concentrate and evaluation of its functional properties, Int. J. Food Sci. Nutr. 58, 486–490.
- Gershenzon J., Croteau R.B. (1993) Terpenoid biosynthesis: the basic pathway and formation of monoterpenes, sesquiterpenes, and diterpenes, in: Moore T.S. (Ed.), Lipid

Metabolism in Plants, CRC Press, Boca Raton, FL, pp. 339–388.

- Heinstein P.F., Herman D.L., Tove S.B., Smith F.H. (1970) Biosynthesis of gossypol: incorporation of mevalonate-2 -14C and isoprenyl pyrophosphates, J. Biol. Chem. 245, 4658–4665.
- Heinstein P.F., Smith F.H., Tove S.B. (1962) Biosynthesis of C14-labeled gossypol, J. Biol. Chem. 237, 2643–2646.
- Keller P.A., Birch C., Leach S.P., Tyssen D., Griffith R. (2003) Novel pharmacophore-based methods reveal gossypol as a reverse transcripterpenoid aldehydese inhibitor, J. Mol. Graph. Model. 21, 365–373.
- Kulkarni V.N., Khadi B.M., Sangam V.S. (2002) Pre-breeding efforts for low gossypol seed and high gossypol plant in G. herbaceum L. cotton utilizing G. australe Mueller, Curr. Sci. India 82, 434–439.
- Lee J.A. (1973) The inheritance of gossypol level in *gossypium* II: Inheritance of seed gossypol in two strains of cultivated *Gossypium barbadense* L., Genetics 75, 259–264.
- Lee J.A. (1978) Allele determining rugate fruit surface in cotton, Crop Sci. 18, 251–254.
- Lee J.A., Cockerham C.C., Smith F.H. (1968) The Inheritance of Gossypol Level in *Gossypium* I. Additive, Dominance, Epistatic, and Maternal Effects Associated with Seed Gossypol in Two Varieties of *Gossypium Hirsutum* L., Genetics 59, 285–298.
- Li Q.F., Cai Y.F., Tan Y.J. et al. (1996) Research on the new cotton sources resistant to many biological strains of Verticillium wilt, Sci. Agr. Sinica, 29, 47–52.
- Lin T.S., Schinazi R.F., Zhu J., Birks E., Carbone R., Si Y., Wu K., Huang L., Prusoff W.H. (1993) Anti-HIV-1 activity and cellular pharmacology of various analogs of gossypol, Biochem. Pharmacol. 46, 251–255.
- Liu J., Benedict C.R., Stipanovic R.D., Bell A.A. (1999) Purification and characterization of S-adenosyl-L-methionine: desoxyhemigossypol-6-O-methyltransferase from cotton plants. An enzyme capable of methylating the defense terpenoids of cotton, Plant Physiol. 121, 1017–1024.
- Liu J., Stipanovic R.D., Bell A.A. (2005) Terpenoid Biosynthetic Pathway in Cotton: Cloning, Expression and Characterization of a Cytochrome P450, Proceedings of the 14th International Conference on Cytochromes, pp. 149–450.
- Liu J., Stipanovic R., Cai Y., Puckhaber L., Bell A. (2007) Cloning and Heterologous Expression of a Cytochrome P450 Hydroxylase Involved in the Terpenoid Biosynthetic Pathway in Cotton. Proceedings-Beltwide Cotton Conferences, USA, Jan 11.
- Liu S., Kulp S.K., Sugimoto Y., Jiang J., Chang H.L., Dowd M.K., Wan P., Lin Y.C. (2002) The (-)-enantiomer of gossypol possesses higher anticancer potency than racemic gossypol in human breast cancer, Anticancer Res. 22, 33–38.
- Lopez L.M., Grimes D.A., Schulz K.F. (2005) Nonhormonal drugs for contraception in men: a systematic review, Obstet. Gyn. Surv. 60, 746–752.
- Luo P., Wang Y.H., Wang G.D., Essenberg M., Chen X.Y. (2001) Molecular cloning and functional identification of (+)-δ-cadinene-8-hydroxylase, a cytochrome P450 monooxygenase (CYP706B1) of cotton sesquiterpene biosynthesis, Plant J. 28, 95–104.

- Lusas E.W., Jividen G.M. (1987) Glandless cottonseed: A review of the first 25 years of processing and utilization research, J. Am. Oil Chem. Soc. 64, 839–854.
- Martin G.S., Jinggao L., Chauncey R.B., Stipanovic R.D., Magill C.W. (2003) Reduced levels of cadinane sesquiterpenoids in cotton plants expressing antisense (+)-δ-cadinene synthase, Phytochemistry 62, 31–38.
- McCarty J.C. Jr., Hedin P.A., Stipanovic R.D. (1996) Cotton Gossypium spp. plant gossypol contents of selected GL2 and GL3 alleles, J. Agr. Food Chem. 44, 613–616.
- McMichael S.C. (1954) Glandless boll in upland cotton and its use in the study of natural crossing, Agron. J. 46, 527–528.
- McMichael S.C. (1959) Hopi cotton, a source of cottonseed free of gossypol pigments, Agron. J. 51, 630.
- McMichael S.C. (1960) Combined effects of the glandless genes gl2 and gl3 on pigment glands in the cotton plant, Agron. J. 52, 385–386.
- Meng Y., Jia J., Liu C., Liang W., Heinstein P., Chen X. (1999) Coordinated accumulation of (+)- delta -cadinene synthase mRNAs and gossypol in developing seeds of *Gossypium hirsutum* and a new member of the cad1 family from G. arboretum, J. Nat. Prod. 62, 248–252.
- Mergeai G. (1992) New perspectives concerning the methodology to be used for introgression of the glanded-plant and glandless-seed character in cultivated cotton (*Gossypium hirsutum* L.), Coton Fibres Trop. 47, 113–119.
- Mergeai G., Baudoin J.P., Vroh Bi I. (1997) Exploitation of trispecific hybrids to introgress the glandless seed and glanded plant trait of *Gossypium sturtianum Willis* into *G. hirsutum* L., Biotechnol. Agron. Soc. Environ. 1, 272–277.
- Mergeai G., Irie V.B., Dujardin P., Baudoin J.P. (1995) Introgression of glanded-plant and glandless-seed trait from G. sturtianum Willis into tetraploid cotton plants, Proceedings Beltwide Cotton Conferences, San Antonio, TX, USA, Jan 1, pp. 513–514.
- Miravalle R.J. (1972) The plant geneticist's contribution toward changing the lipid and amino acid composition of cottonseed, J. Am. Oil Chem. Soc. 49, 24–26.
- Oliver C.L., Miranda M.B., Shangary S., Land S., Wang S., Johnson D.E. (2005) (–)-Gossypol acts directly on the mitochondria to overcome Bcl-2- and Bcl-X(L)-mediated apoptosis resistance, Mol. Cancer Ther. 4, 23–31.
- Punit M., Singh P., Dongre A.B., Narayanan S.S. (1995) Gossypol-gland density and free gossypol content in seed and cotyledonary leaf of upland cotton (*Gossypium hirsutum*), J. Agr. Sci. 65, 66–68.
- Punit M., Singh P., Narayanan S.S. (1991) Variability for gossypol glands in upland cotton (*Gossypium hirsutum* L.), Source Adv. Plant Sci. 4, 165–170.
- Singh I.D., Weaver J.B. (1972) Growth and infestation of boll weevils on normal-glanded, grandless, and high-gossypol strains of cotton, J. Econ. Entomol. 65, 821–824.
- Singh P., Singh T.H., Chahal G.S. (1991) Genetical control of gossypol content in intervarietal crosses of upland cotton, Crop Improv. 18, 141–143.
- Smith F.H. (1962) Synthesis and translocation of gossypol by cotton plant, Proc Beltwide Cotton Prod. Res. Conf., pp. 7–12.

- Stipanovic R.D. (1992) Natural product biosynthesis via the Diels-Alder reaction, in: Petroski R.J., McCormick S.P. (Eds.), Secondary-Metabolites and Metabolism, Plenum Press, New York, pp. 319–328.
- Stipanovic R.D., Bell A.A., Benedict C.R. (1999) Cotton pest resistance: the role of pigment gland constituents, in: Cutler H.G., Cutler S. (Eds.), Biologically Active Natural Products: Agrochemicals, CRC Press, Boca Raton, Florida, pp. 211–220.
- Stipanovic R.D., Bell A.A., Lukefahr M.J. (1977) Natural insecticides in cotton (*Gossypium*), in: Hedin P.A. (Ed.), Chemical Basis.
- Stipanovic R.D., Stoessl A., Stothers J.B., Altman D.W., Bell A.A., Heinstein P. (1986) The stereochemistry of the biosynthetic precursor of gossypol, J. Chem. Soc. Chem. Comm. 2, 100–102.
- Sunilkumar G., Campbell L.M., Puckhaber L., Stipanoic R.D., Rathore K.S. (2006) Engineering cottonseed for use in human nutrition by tissue-specific reduction of toxic gossypol, PNAS 103, 18054–18059.
- Tan X.-P., Liang W.-Q., Liu C.-J., Luo P., Heinstein P., Chen X.-Y. (2000) Expression pattern of (+)-delta-cadinene synthase genes and biosynthesis of sesquiterpene aldehydes in plants of *Gossypium arboreum* L., Planta 210, 644–651.
- Tegos G., Stermitz F.R., Lomovskaya O., Lewis K. (2002) Multidrug pump inhibitors uncover remarkable activity of plant antimicrobials, Antimicrob. Agents Ch. 46, 3133–3141.
- Townsend B.J., Poole A., Blake C.J., Llewellyn D.J. (2005) Antisense suppression of a (+)- delta -cadinene synthase gene in cotton prevents the induction of this defense response gene during bacterial blight infection but not its constitutive expression, Plant Physiol. 138, 516–528.
- Vaissayre M., Hau B. (1985) New results on the susceptibility of glandless cotton varieties to phyllophagous insects, Coton Fibres Trop. 40, 159–168.
- Vroh Bi I., Maquet A., Baudoin J.P., Jardin P., Jacquemin J.M., Mergeai G. (1999) Breeding for "low-gossypol seed and high-gossypol plants" in upland cotton, Analysis of trispecies hybrids and backcross progenies using AFLPs and mapped RFLPs, Theor. Appl. Genet. 99, 1233–1244.
- Wang J.Y., Cai Y., Gou J.Y., Mao Y.B., Xu Y.H., Jiang W.H., Chen X.Y. (2004) VdNEP, an Elicitor from *Verticillium dahliae*, Induces Cotton Plant Wilting, Appl. Environ. Microb. 70, 4989–4995.
- Wilson F.D., Smith J.N. (1976) Some genetic relationships between gland density and gossypol content in *Gossypium hirsutum* L., Crop Sci. 16, 830–832.
- Xie Yong-Fang, Li Biao, Cai Ying-Fan, Wang Bochu (2007) Construction of cDNA Library of Cotton Mutant (Xiangmian 18) Library During Gland Forming Stage, Colloid. Surface. B: Biointerfaces 60, 258–263.
- Xu YanHua, Wang JiaWei, Wang Shui, Wang JianYing, Chen XiaoYa, Characterization of GaWRKY1, a cotton transcription factor that regulates the sesquiterpene synthase gene (+)- delta -cadinene synthase-A, Plant Physiol. 135, 507–515.
- Yan-Hong W., Guadalupe D.-H., Essenberg M. (2003) 8-Hydroxy-(+)-delta-cadinene is a precursor to hemigossypol in *Gossypium hirsutum*, Phytochemistry 63, 219–225.
- Ye W., Chang H.L., Wang L.S., Huang Y.W., Shu S., Dowd M.K., Wan P.J., Sugimoto Y., Lin Y.C. (2007) Modula-

tion of multidrug resistance gene expression in human breast cancer cells by (–)-gossypol-enriched cottonseed oil, Anticancer Res. 27, 107–116.

- Zhang X., Jin L., Zhang T. (2001) A new upland cotton cultivar with glanded plant and low gossypol content seed, Sci. Agr. Sinica 34, 564–567.
- Zhang T.-Z., Zhang X.-L., Jin L. (2001) Genetic Identification of a New Gland Forming Gene in Upland Cotton, Acta Agron. Sinica 27, 75–79.
- Zhu S.J., Li B.L. (1993) Studies of introgression of the "glandless seeds-glanded plant" trait from Gossypium bickii into cultivated Upland cotton (Gossypium hirsutum), Source Coton et Fibres Tropicales 48, 195–200.
- Zhu S., Ji D., Wang R., Wang H. (1999) Studies on the gossypol trend of the cotyledon during seed germination and the relationship between gossypol and gland formation in the wild species of *Gossypium* in Australia, Acta Gossypii Sinica 11, 169–173.
- Zhu S., Jiang Y., Reddy N., Daofan J.I. (2004) Breeding, introgression and inheritance of delayed gland morphogenesis trait from *Gosspium bickii* into upland cotton germplasm, Chinese Sci. Bull. 9, 2470–2476.
- Zhu S.J., Reddy N., Jiang Y.R. (2005) Introgression of a gene for delayed pigment gland morphogenesis from *Gossypium bickii* into upland cotton, Plant Breeding 124, 591–594.

# Micronutrient-Efficient Genotypes for Crop Yield and Nutritional Quality in Sustainable Agriculture

Amir Hossein Khoshgoftarmanesh, Rainer Schulin, Rufus L. Chaney, Bahareh Daneshbakhsh, and Majid Afyuni

### Contents

1	Intro	duction	220			
2	Impr	oving Crop Yield	221			
3	Crop	Quality and Human Health	222			
	3.1	Consequences of Micronutrient Deficiencies	222			
	3.2	Nutrient Bioavailability	223			
4	Corr	ection of Micronutrient Deficiency	223			
	4.1	Fertilizer Application	223			
	4.2	Soil Amendments	224			
	4.3	Organic Fertilizer Sources	225			
	4.4	Plant Residue	226			
	4.5	Animal Manure	226			
	4.6	Organic Amendments	227			
	4.7	Crop Rotation	228			
5	Deve	Developing Micronutrient-Efficient Genotypes:				
	A Ne	w Strategy	229			
	5.1	Definition of Nutrient Efficiency	229			
	5.2	Micronutrient-Efficient Genotypes and Crop				
		Productivity	229			
	5.3	Micronutrient-Efficient Genotypes, Crop				
		Quality and Human Health	231			
	5.4	Combining the Benefits of High Yield and				
		Better Nutritional Quality	234			
	5.5	Need to Limit Grain Cd and As While				
		Improving Levels of Zn and Fe in Crops	235			
	5.6	Challenges in Programs for Selecting or				
		Breeding Micronutrient-Efficient Genotypes	236			
	5.7	Stress Tolerance Indicators for Selecting				
		Micronutrient-Efficient Genotypes	238			
6	Conc	lusion	239			
Re	eferenc	es	243			

A.H. Khoshgoftarmanesh (⊠)

Department of Soil Science, Isfahan University of Technology, 84154 Isfahan, Iran e-mail: amirhkhosh@cc.iut.ac.ir Abstract About 4 billion people will be added onto the present population by 2050. To meet further demand for food, agricultural production should increase on the existing land. Since the Green Revolution, higher crop production per unit area has resulted in greater depletion of soil phytoavailable micronutrients while less attention has been paid to micronutrients fertilization. Now, micronutrient deficiency has become a limiting factor for crop productivity in many agricultural lands worldwide. Furthermore, many food systems in developing countries can not provide sufficient micronutrient content to meet the demands of their citizens, especially low-income families. There are several solutions such as soil and foliar fertilization, crop systems, application of organic amendments to correct micronutrients deficiency and to increase their density in edible parts of plants. This review article presents (1) agronomic approaches to improve crop yield and micronutrient content of food crops, and (2) genotypic variation in uptake and accumulation of micronutrients. Considering ecological concerns, cultivation and breeding of micronutrient-efficient genotypes in combination with proper agronomic management practices appear as the most sustainable and cost-effective solution for alleviating food-chain micronutrient deficiency. Micronutrient-efficient genotypes could provide a number of benefits such as reductions in the use of fertilizers, improvements in seedling vigor, and resistance to biotic and abiotic stresses. Using bioavailable micronutrientdense staple crop cultivars can also be used to improve the micronutrient nutritional status of human.

**Keywords** Micronutrients • Nutrient efficiency • Biofortification • Stress-tolerance indicators

#### **1** Introduction

The world's population is estimated to increase from 6 billion to about 10 billion by 2050. To meet the food demand of the growing world population, a large increase in food production is required. At the same time, the increases in world population will result in serious pressure on the existing agricultural land through urbanization and intensive cultivation (Alexandratos, 1995; Byrnes and Bumb, 1998). It has been estimated that to supply enough food for the world population in 2020, annual cereal production needs to increase by 40%, from 1773 billion tons in 1993 to nearly 2500 billion tons in 2020 (Rosegrant et al., 1999, 2001). About 85% of the increase in total cereal demand will occur in the developing countries.

There are two strategies to increase food production: (a) to expand the agricultural area and (b) to enhance crop yields per unit area. The expansion of agricultural area is limited due to lack of suitable lands, urbanization, degradation of soils, and increasing water scarcity (Brown, 1997). To meet the further demand for food, agricultural production must increase on the existing land. This means that more crop food production must be achieved per unit of the presently available arable land.

Since the "Green Revolution", intensive cropping, cultivation of high-yield genotypes, improved agricultural mechanization, production of macronutrient fertilizers with low impurities of trace elements, and using modern irrigation systems has resulted in higher crop production per unit area and greater depletion of soil phytoavailable micronutrients.

Losses of micronutrients through erosion, leaching, liming of acid soils, decreased proportions of farmyard manure compared to chemical fertilizers, and use of marginal lands for crop production are other factors that have increased the incidence of micronutrient deficiencies in agricultural soil worldwide (Fageria et al., 2002). Micronutrient deficiency problems are also exacerbated by the high demand of modern crop cultivars. Accordingly, low levels of micronutrients have been reported on many crops grown in different countries (Cakmak et al., 1996; Fageria, 2000a; Galrão, 1999; Graham et al., 1992; Grewal and Graham, 1999; Martens and Lindsay, 1990). Therefore, micronutrient deficiency has become a limiting factor for crop productivity in many agricultural soils. In order to obtain the genetic potential yields of crops, correcting micronutrients deficiencies is necessary.

On the other hand, agricultural practices have almost always targeted higher crop yield production while minimizing costs. In addition, nutrient output of farming systems has never been a goal of either agriculture or of public policy. Thus, the increase in crop yield in many agricultural systems as a result of chemical fertilizer application has been accompanied with reduced micronutrients concentrations in the edible parts of different crops. So that, already, many food systems in developing countries can not provide sufficient micronutrients concentrations to meet the demands of their people especially low-income families (ACC/SCN, 1992, 1997, 2000; Combs et al., 1996; Combs and Welch, 1998; McIntyre et al., 2001; Welch et al. 1997; Welch, 1998).

Recently, mid 1980s, the nutrition community reported the global significance of micronutrient malnutrition (Allen, 2000). In 2001, fighting micronutrient malnutrition was considered an integral component of three of the eight Millennium Development Goals (MDGs) of the General Assembly of the UN to alleviate the world's greatest health and poverty issues by 2015 (United Nations General Assembly, 2000). The focus of improving the micronutrient quality of crops notes the density of bioavailable micronutrients in crops as consumed in order to take into account crop factors which increase or decrease the bioavailability of crop micronutrients (see below). For example, phytate in grains is well known to inhibit absorption or utilization of crop Zn by humans and livestock, such that grain phytate must be considered whenever the quality of grain Zn for human use is considered.

Deficiencies of iron, zinc, vitamin A, and iodine not only compromise the immune system, but can irreversibly retard development in infancy. Deficiency of any of these in a pregnant or lactating woman can result in subclinical mental retardation in children. Such children may be less fit to control their environment and to provide for their own food security in later life, to compete for better education and for higher level jobs within their society. Vitamin A deficiency can lead to poor night vision, eye lesions and, in severe cases, permanent blindness; increased illness and death from infections. Iodine deficiency can cause goiter, mental retardation, brain damage and reproductive failure. Iron deficiency can cause nutritional anemia, problem pregnancies, stunted growth, and lower resistance to infections, long-term impairment in mental function, decreased productivity, and impaired neural development. Zinc deficiency can

cause growth retardation, delayed skeletal and sexual maturity, dermatitis, diarrhoea, alopecia, and defects in immune function with resulting increase in susceptibility to infection.

Several approaches such as soil and foliar fertilization, improved crop systems, application of soil amendments and organic nutrient sources are possible to correct micronutrients deficiency and to increase their density in edible parts of plants. Considering ecological concerns, sustainable agriculture is looking for more environmental-friendly and cost effective approaches with low input of energy and chemicals. Among different strategies applied for correcting micronutrient deficiency in the food chain from plant to human, the most sustainable solution, particularly for developing countries, seems to be cultivation and breeding of micronutrient-efficient genotypes in combination with proper agronomic management practices.

Nutrient efficiency and nutrient-efficient genotypes have been defined in different ways (Stewart et al., 2005; Fageria et al., 2008). Definitions of nutrient efficiency generally can be divided into those emphasizing crop productivity and those emphasizing the internal plant nutrient requirement (Gourley et al., 1994). With regard to yield parameters, nutrient efficiency has been defined as the ability to produce a high plant yield in a soil that would otherwise limit the production of a standard genotype (Marschner, 1995). Other definitions of nutrient efficiency also referred to as 'agronomic efficiency', include the amount of shoot production per unit of nutrient applied (Caradus, 1990; Sauerbeck and Helal, 1990). Nutrient efficiency has also defined as yield response per unit of added nutrient or of nutrient in the crop shoots (Baligar et al., 1990; Blair, 1993; Thung, 1988). This paper will briefly discuss some aspects of using and developing micronutrient-efficient crop plant genotypes as a sustainable and cost-effective method for alleviating food-chain micronutrient deficiency malnutrition of humans.

### 2 Improving Crop Yield

Mineral nutrients have played an important role in enhancing crop production since the beginning of the green revolution. According to Borlaug and Dowswell (1993), nearly half of the increase in crop yields during the 20th century was due to chemical fertilizer application. Generally, improving the nutritional status of plants by applying fertilizers and maintaining soil fertility has been the critical step in doubling food production both in developed and developing countries (Loneragen, 1997). The increase in crop production has been mainly resulted from the application of N-P-K fertilizers (Brady and Weil, 2002). Since 1950, application of N-P-K fertilizers has been rapidly increased in agricultural lands while less attention has been paid to micronutrients fertilization. Further, higher crop production per unit area depleted greater amount of micronutrients from soils. Now, micronutrient deficiency is a limiting factor for attaining greater crop yields in many cultivated soils.

Accordingly, micronutrient deficiency has been reported widely in plant crops grown in different countries. For example, Fe deficiency in crops have been reported for various regions of Europe, east India, Bangladesh, Malta, Turkey, Zambia, Mexico (Sillanpaa, 1982), Indonesia (Katyal and Vlek, 1985; Welch et al., 1991), several North, Central and South American countries (Leon et al., 1985), Australia, and most Mediterranean and west African countries (Donald and Prescott, 1975).

Zinc-deficient soils are widespread in Mediterranean countries (Cakmak et al., 1997b; Khoshgoftarmanesh et al., 2004a), Australia (Donald and Prescott, 1975; Sillanpaa and Vlek, 1985), China (Takkar and Walker, 1993), Africa (Cottenie et al., 1981), Asia (Katyal and Vlek, 1985; Welch et al., 1991), and North and South America (Fageria, 2000b; Lopes and Cox, 1977). Some crops (maize, rice) are especially susceptible to Zn deficiency in most countries where they are grown.

Manganese deficiency has also been reported in soils of the United States (Reuter et al., 1988), Central and South America (Leon et al., 1985), Europe (Welch et al., 1991), semiarid regions of China, India, southeast and western Australia, and many African countries (Welch et al., 1991).

The importance of certain trace elements for optimum growth and development of plants has been reported for the last century (Barker and Pilbeam, 2007; Cakmak, 2002; Marschner, 1995). Up to now, several trace elements (B, Br, Cl, Co, Cu, F, Fe, I, Mn, Mo, Ni, Na, Rb, Si, V, and Zn) are known to be essential for certain plants. Several nutrients have been shown necessary only for a few species (Al, I, Na, Si, Rb, Ti, V, F, Br), and others are known to have stimulating effects on plant growth, although their functions are not yet recognized (Barker and Pilbeam, 2007). The essential micronutrients are those that their specific biochemical roles cannot be substituted by other elements. In the absence of adequate amounts of these elements inside plant tissues, crop growth and metabolic cycles will be impaired.

Several studies have reported the significant reduction in growth and yield of different plant species in soils deficient in Fe, Mn, Zn, Cu, B, and other micronutrients (Cakmak et al., 1998; Kalayci et al., 1999; Khoshgoftarmanesh et al., 2004a, b, 2006a, b, 2007; Khoshgoftar and HajiMozaffari, 2006). Therefore, correcting micronutrient deficiency through appropriate approaches is necessary to achieve higher crop yields.

#### 3 Crop Quality and Human Health

The increase in food production to prevent starvation and malnutrition was the main goal driving the agricultural "Green Revolution" during the latter half of the 20th century (Welch and Graham, 1999). In parallel to increase in crop yield in many agricultural systems as a result of chemical fertilizer application and improved agronomic practices, micronutrients contents in the edible parts of different crops were reduced to levels lower than that of the traditional crops. Thus, many food systems in developing countries can not provide sufficient micronutrient contents to meet the demands of their people especially low-income families (ACC/SCN, 1992, 1997, 2000; Combs et al., 1996; Combs and Welch, 1998; McIntyre et al., 2001; Welch et al., 1997; Welch, 1998). Humans need trace elements in sufficient levels to sustain life. There are also some dietary substances such as phytate that affect nutrients bioavailability and as a result, human nutritional status (ACC/SCN, 1992, 1997, 2000; Combs et al., 1996; Combs and Welch, 1998; McGuire, 1993; Welch et al., 1997; Welch, 1998).

Micronutrient malnutrition or hidden hunger now affects more than 40% of the world's population (United Nations Administrative Committee on Coordination – Subcommittee on Nutrition, 1992). The consequences of micronutrient deficiencies are mostly found among women, infants and children from poor families in developing countries (Buyckx, 1993; Ramalingaswami, 1995; World Health Organization, 1992; Mason and Garcia, 1993). However, micronutrient malnutrition is not restricted to developing countries and is also widespread in well-developed industrialized countries (Cobiac and Baghurst, 1993; US Department of Health and Human Services, 1990).

There has been a huge increase in the number of people suffering from micronutrient malnutrition over the last four decades. For example, Fe deficiency anemia has grown from about 30% of the world's population in the 1960s to over 40% during the late 1990s. It has been estimated that over 3.7 and 1 billion people are at the risk of iron and iodine deficiency disorders, respectively. In addition, Zn, Se, vitamin A, vitamin C, vitamin D and folic acid deficiencies are widespread among populations especially in developing countries (Combs et al., 1996; World Health Organization, 1999). Many zinc nutrition specialists believe that zinc is as important as iron deficiency (Gibson, 1994; Gibson et al., 2008). Copper, boron, manganese, chromium and lithium are other micronutrients that are needed for nutrient balance.

# 3.1 Consequences of Micronutrient Deficiencies

Impaired immune function, increased mortality and morbidity rates, lower worker productivity, diminished intellectual performance, lower educational attainment, a lower livelihood, higher birth rates, lower standard of living performance, and increased rates of chronic diseases including coronary heart disease, cancer and diabetes are some consequences of micronutrient hidden hunger (Pinstrup-Andersen, 1999; Sanghvi, 1996; Welch and Graham, 2000).

Micronutrients malnutrition increase human susceptibility to infections and is the most important risk factor for the disease in developing countries (Brabin and Coulter, 2003; Food and Agriculture Organization of the United Nations, 2004; Murray and Lopez, 1997; Black, 2003). Anemia, problem pregnancies, stunted growth, lower resistance to infections, long term impairment in mental function, decreased productivity and impaired neural development are some consequences of iron deficiency. Zinc deficiency may result in growth retardation, delayed skeletal and sexual maturity, dermatitis, diarrhea, alopecia and defects in immune function with resulting increase in susceptibility to infection (Pinstrup-Andersen, 1999; Welch and Graham, 2000).

#### 3.2 Nutrient Bioavailability

The nutritional quality of a diet can be determined based on the concentration of individual nutrients as well as interactions of other elements, promoters and antinutrients which affect bioavailability of micronutrients. In fact, the degree to which a nutrient is absorbed from the diet depends on antinutrients contents e.g., phytate, tannins and antivitamins (Graham et al., 2000). High levels of antinutrients including phytate and tannins in the diet reduce the bioavailability of micronutrients including Fe and Zn, while promoters such as vitamin A or  $\beta$ -carotene and vitamin C enhance micronutrients bioavailability in humans (Garcia-Casal et al., 1998). Phytate is an important antinutrient to zinc absorption. Phytate is especially high in cereal grains. Depending on the plant genotype, about 70 to 90 % of the total phosphorus in whole grain is in the form of phytate (O'Dell et al., 1972). Other antinutrients are of lesser importance in cereals.

Another factor affecting the bioavailability of micronutrients is interaction with other nutrients. For example, interactions between vitamin A (retinol) and zinc has been found in the late 20th century (Christian and West, 1998; Solomons and Russell, 1980). Syner-gistic and antagonistic interactions among chemically similar metal cations have been characterized in both plants and animals (Hill and Matrone, 1970). Such a synergistic effect strongly indicates that breeding for staples that are dense in both iron and zinc is required in order to effectively address iron-deficiency anemia. It has been reported that the anemia could be corrected using vitamin A but not iron supplementation (Hodges et al., 1978).

Cereal-based foods with low concentrations and reduced bioavailability of micronutrients have been considered a major reason for the widespread deficiencies of Zn and Fe in developing countries (Graham et al., 2001; Welch and Graham, 1999). Thus it is essential to find sustainable approaches to produce food supplies adequate in quantity and proper nutritional quality.

#### 4 Correction of Micronutrient Deficiency

#### 4.1 Fertilizer Application

Alleviation of micronutrient deficiencies to improve crop yields is difficult due to large temporal and spatial variation in phytoavailability of soil micronutrients (Brennan and Bolland 2006; Shaver et al., 2007). The easiest and most straightforward practice to correct micronutrient deficiency is to apply micronutrient fertilizers. Soil application of fertilizer is economically more affordable. However, in soils with high pH, most of soil-applied micronutrients will become unavailable for plant uptake within a few years (Martens and Westerman, 1991). The loss in residual effect of soil-applied nutrient requires adding more fertilizer to maintain a sufficient level of phytoavailable element, although total soil element contents are still high and actually increasing from application to application. Increased total soil metal may create problems of micronutrients losses or toxicity in the future, especially under conditions that mobilize soil metal such as soil acidification. Furthermore, soil application of fertilizers is not always a useful approach to increase micronutrients concentrations in cereal grain. Many studies reported minimal or no increases in grain micronutrient content, with increasing soil application of micronutrient fertilizers even when yield was increased.

Applied soluble micronutrients fertilizers become ineffective rather rapidly as the dissolved metals react with soil minerals and organic matter. In comparison with inorganic fertilizers, synthetic and natural chelates have the advantage of keeping the applied nutrient in solution in a less reactive form. Chelates are particularly appropriate for applications of Zn and Fe to alkaline and calcareous soils. Soil application of inorganic Fe fertilizers to Fe-deficient soils is usually ineffective because of rapid conversion of soluble Fe into plant-unavailable solid Fe(III) forms. In contrast, an application of synthetic Fe-chelates for correction of Fe deficiency is effective (Wallace and Wallace, 1982) for a longer period, or actually long enough to support one crop of tree fruits which are commonly fertilized with Fe-chelates.

The effectiveness of various synthetic and natural chelates has been widely investigated (Alvarez and Gonzalez, 2006; Gonzalez et al., 2007; Prasad and Sinha, 1981). Despite their effectiveness, application of chelates is generally expensive (Wallace and Wallace, 1982), and is likely to be even more uneconomic if the aim is to increase micronutrient concentration in the grain rather than to increase yield. In addition, application of metal chelates may result in potential leaching risk because the more mobile the chelate, or the less biodegradable the carrier, the greater the risk for leaching (Gonzalez et al., 2007).

Foliar application can provide a rapid correction of severe deficiencies commonly found during the early stages of growth, and are temporary solutions to the problem. Foliar fertilization is more effective than soil application, but too expensive for resource-poor farmers. Low penetration rates in thick leaves, run-off from hydrophobic surfaces, wash off by rainfall, rapid drying of spray solution, limited translocation within the plant, and leaf damage are other problems related to foliar applications (Marschner, 1995). Most foliarapplied micronutrients are not efficiently transported toward roots, which may remain deficient. Therefore, application of micronutrients to either soil or foliage may be successful in increasing grain yield only in well-defined circumstances.

Another method to apply micronutrients to crop plants is the treatment of seeds with fertilizers, a technique called "seed priming". The most effective way to prime micronutrients in seeds is to grow the parent crop on a soil with high levels of phytoavailable Mn or Zn, while spraying high levels of Zn and Mn fertilizers on the plants during seed filling can also increase seed levels considerably. Mixing micronutrients with seeds must be limited to levels which do not harm the germinating seeds. It has been reported that seed priming with Zn sulphate resulted in a stronger increase in the grain yield of wheat grown on Zn-deficient soil than foliar Zn application after germination, but it did not significantly increase grain Zn concentration in contrast to foliar and soil application (Rengel and Graham, 1995; Yilmaz et al., 1997, 1998). 'On-farm' seed priming with water is an effective way to increase yield in maize and is important for resource-poor farmers as it is a low- or zero-cost technology and requires few external inputs (Harris et al., 2007). Recently, Harris et al. (2008) reported that priming seeds with ZnSO<sub>4</sub> significantly increased the grain Zn concentration and concluded that using ZnSO<sub>4</sub> to prime seeds was very cost-effective, with net benefit-to-cost ratios of 75 for wheat and 780 for chickpea.

Correction of micronutrients deficiency via fertilization, especially in calcareous soils, is not always successful and sustainable due to agronomic and economic factors such as reduced availability of micronutrient element due to topsoil drying, subsoil constraints, disease interactions, and cost of fertilizer in developing countries (Graham and Rengel, 1993).

Some chemical fertilizers, especially Zn and Fe fertilizer manufactured from industrial byproducts or mine wastes contain various contaminants such as cadmium and Pb (Afyuni et al., 2007). The potential risks to environmental quality posed by their repeated application to soil require careful evaluation (Mortvedt, 1985, 1996) especially for rice production (Chaney et al., 2004). Recently, preserving the environment is becoming a more important objective of agriculture to achieve sustainable agricultural goals (Cakmak, 2002; Tillman, 1999). To avoid potential future environmental risks, a more sustainable strategy is required than just adding chemical micronutrients fertilizers to the soil. Total amounts of some micronutrients such as Zn and Fe in agricultural soils are generally orders of magnitude higher than the crop demand. Thus, micronutrients deficiency in crops is a problem of insufficient phytoavailability in almost all cases and not of absolute lacking sufficient quantity of micronutrients.

#### 4.2 Soil Amendments

Soil amendments are frequently used by farmers to improve some physical and chemical properties of soils (e.g. adjusting soil pH) in order to enhance plant growth. Limestone application is an effective and common practice to improve crop productivity of acidic soils (Foy, 1984, 1992; Kochian, 1995). Adding limestone may depress the uptake of Zn, Cu, Fe, and Co, and increase the uptake of Se and Mo by plants (Foy, 1984). The pH largely controls micronutrients solubility in soil and a small change in pH significantly affects solubility and uptake of metals such as Zn, Cu, Mn, and Fe. Considering increased pH of acidic soils by liming, micronutrient deficiencies become more severe after liming (Verma and Minhas, 1987). A high soil-pH favors the oxidation of reduced forms of Se such as  $Se^{-2}$  and  $SeO_3^{-2}$  to the more soluble and plantavailable  $\text{SeO}_4^{-2}$  anion.

Several studies have examined the effects of combined fertilization with lime and P. Phosphorus causes changes in the extractability of Zn and, in almost all soils, further aggravates Zn deficiency in susceptible crops causing decreases in grain Zn concentration (Hylander, 1995). Thus, Zn fertilizers must often be applied to ameliorate Zn deficiency associated with limestone application (Tagwira, 1993).

In contrast to limestone, gypsum (CaSO<sub>4</sub>) and elemental S are used to decrease the pH of alkaline soils as well as to amend sodic and saline-sodic soils. Application of acid-producing amendments on alkaline and calcareous soils could decrease soil pH and consequently increase plant-available Fe, Mn, Zn, Cu, and Co. Gypsum is used to exchange Ca for Na on the soil cation-exchange complex and to remove bicarbonate from the soil solution. Removal of bicarbonate from the soil solution can be highly beneficial for lowering the soil pH and increasing the availability of micronutrients (Singh et al., 1989). Singh et al. (1989) reported that gypsum application in a sodic soil increased the mean concentration of Mn, Zn and Cu in mesquite (Prosopis julitlora). Application of gypsum caused significant reductions in pH and electrical conductivity (Singh et al., 1989). The increase in soil availability of micronutrient as a result of gypsum application might be attributed to the decrease in soil pH and improved soil physical properties. Further research is necessary to ascertain whether benefits of applying gypsum to alkali, sodic soils are significant in terms of increasing micronutrient density in grain.

### 4.3 Organic Fertilizer Sources

Application of different organic materials e.g. manures, plant residues and waste materials is a recommended strategy to improve soil fertility and enhance nutrient use efficiency. Organic sources application is known as an effective management strategy to sustain high crop productivity in many cropping systems. Using organic sources may affect crop micronutrient nutrition through affecting soil physical and chemical properties, producing better root growth environment and adding some micronutrients to soil (Rengel et al., 1999).

Soil organic matter has a variety of direct and indirect influences on the phytoavailability of micronutrients in soil and their uptake by plants (Rengel et al., 1999). Binding of some metals such as Zn to organic matter reduces the concentration of free cation in solution, but if the metal-organic complexes are dissolved, then the increase in total dissolved ion may actually also increase effective metal phytoavailability at the root-rhizosphere interface, depending on the mobility and the dissociation kinetics of the metal-DOC complexes. Chelation of Zn and Fe by organic matter holds these nutrients in forms more accessible by roots within the rhizosphere, and retards formation of insoluble solid forms such as oxides and carbonates in soil (Schulin et al., 2009).

Adding organic resources such as crop residues, green manure, livestock manure, municipal biosolids or their composts or co-composts to soil has a number of beneficial effects on micronutrient nutrition including additional supply of some nutrients with the added organic matter, increase in ion exchange capacity and thus of the fractions of easily available nutrients, improved soil structure, increase in water storage capacity, improved drainage and aeration, increased accessibility of soil for plant roots, decreased salinity, stimulation of microbial exudation of organic ligands and of other microbial activities and as a result, accelerated release of nutrients from secondary soil matter, enhanced nutrient supply through mycorrhizae, and protection against root pathogens. All these effects promote plant growth, the development of the root system and thus also its capacity to acquire micronutrients (Schulin et al., 2009).

The formation of metal complexes with organic ligands in the soil solution is generally found to increase the mobility of soil micronutrients. On the other hand chelation can substantially reduce cellular metal uptake. Because microelements bound to the soil surfaces can only reach roots by diffusion, the mobilization effect of chelation generally seems to cause greater increase in plant uptake. In previous studies even strong chelates (Zn-EDDS, Zn-EDTA) still resulted in a net increase of Zn uptake (Novack et al., 2008). This can be explained by increased rate of movement of the microelement from soil surfaces to root uptake surfaces by movement of dissolved chelated elements (e.g., Wallace and Mueller, 1973). It has also to be considered

that leaf-applied chelates might not be helpful for transport of micronutrients into grain (Cakmak, 2008; Novack et al., 2008).

Effects of different organic amendments on correcting micronutrients deficiencies have been widely investigated; however, very few controlled experiments have been done to determine which types of organic matter practices significantly enhance or depress the levels of micronutrients in edible portions of major food crops. Therefore, more research is needed to understand the impact of various types of organic amendments on crop nutritional quality.

#### 4.4 Plant Residue

In many countries, especially arid and semi-arid regions, very little or no crop residue is left in the field; much is used for feeding animals or for fuel (Timsina and Connor, 2001). It is estimated that more than 1000 million tons of cereal residues are being annually produced in the developing countries (FAO, 1999). Rice, wheat, corn, soybean, barley, rapeseed, and potato are the major residue-producing crops that globally produced 2956 million tons of residues in 1998 (FAO, 1999).

Crop residues are considered as important sources of several micronutrients. Rice and wheat remove 96, 777, 745, 42, 55, and 4 g ha<sup>-1</sup> of Zn, Fe, Mn, Cu, B, and Mo per ton, respectively. By estimating the total crop residue production of 105 million tons in India, and based on micronutrient contents of the residues, the micronutrient potential associated with crop residues would be about 35400 tons (Prasad, 1999). About 50 to 80% of Zn, Cu, and Mn taken up by rice and wheat crops can be recycled through residue incorporation (Prasad and Sinha, 1995b). Therefore, recycling of crop residues can help improve soil availability of micronutrients.

Crop residues addition to flooded soils stimulates microbial metabolism and thus increases soil solution Fe and Mn concentrations due to greater redox potential change (Katyal, 1977; Yodkeaw and De Datta, 1989; Atta et al., 1996). In calcareous soils, organic acids produced during crop residue decomposition may increase plant Zn uptake by dissolving Zn from the solid-phase pool to soil solution (Prasad and Sinha, 1995a). Chelating agents released from decomposing crop residues increase the concentration of total diffusible Zn and its diffusion coefficient (Singh et al., 2005). Rice straw application has been found to increase the Zn content of rice plants, possibly through its amelioration of soil pH and exchangeable sodium percentage (Singh et al., 2005). In contrast, crop residue application has been reported to decrease the labile Zn pools in an alkaline soil (Kang, 1988). Other researchers (Raj and Gupta, 1986) have also reported that application of rice or wheat straw decreased the phytoavailable Zn concentration in both flooded and upland soils.

Crop residues may also negatively affect crop production in the short term because of N immobilization and phytotoxic compounds release in allelopathic interactions. Plants may release chemical compounds which can either stimulate or inhibit the growth and development of another plant. These compounds are dispersed into the environment from root exudates, leaf leachates, or from dead and decaying plant parts (Rose et al., 1984; Huber and Abney, 1986). Nutrient uptake may be reduced by allelochemicals (Bhowmik and Doll, 1984; Rice, 1984; Barnes and Putnam, 1986). It has been suggested that sweet potato residues have an allelopathic effect (Walker and Jenkins, 1986). Sweet potato and cowpea indicator plants grown in sweet potato crop residues displayed symptoms resembling complex nutrient deficiencies. Abiotic factors, such as reduced soil pH, increased soil osmotic potential, reduced oxygen concentrations and low nutrient content of the residue, may also have caused growth inhibition and as a result, reduced plant nutrient uptake. Walker et al. (1989) reported that plant uptake of P, K, Ca, Mg, S, Zn, B and Cu were substantially lower in plants grown in soil with amendment by sweet potato root or vine residues than in plants grown in without sweet potato residue treatments. Successful integration of crop residue management strategies into cropping systems requires understanding of how crop residues influence cycling of nutrients from soil and fertilizers, as well as their effects on soil chemical, physical and biological properties, and crop production.

### 4.5 Animal Manure

Livestock manure is a good source of plant nutrients and its application in agricultural lands is a traditional nutrient management in many countries worldwide. The use of manures can also change plant-available micronutrients by adding micronutrients or by changing both the physical and biological characteristics of the soil (Stevenson, 1991, 1994). Eghball et al. (2004) found that applications of manure and compost not only improved soil physical properties but also provided nutrients for growing corn. The residual effects of manure on soil properties remained several years after application. Gao et al. (2000) found that manure was a better source of available Fe, Mn, and Zn compared to synthetic fertilizers, but manure accelerated the depletion of available Cu. Manure not only supplies large amounts of Zn to the soil, but also promotes biological and chemical reactions that result in the dissolution of non-available Zn (Wei et al., 2006). Wei et al. (2006) reported that manure application had no significant effect on available Cu. This might be attributed to the low amount of Cu in the manure used in this experiment. Copper could be bound to organic matter and thus be relatively unavailable to plants.

Zinc and Cu may be added to poultry feeds as a preventive for scour and to suppress bacterial action in the gut. Accordingly, Zn and Cu concentrations of some compound poultry and swine feeds are several times higher than that required by the animals for healthy development. In areas where animal manures have been applied for many years and where applications are expected to continue, high amounts of heavy metals (particularly Zn and Cu) may accumulate in soil (Nicholson et al., 1999). The micronutrient contents of farm manures largely depend on their concentrations in the feeds consumed and the efficiency of feed conversion by the animals. Fleming and Mordenti (1991) reported mean Zn concentrations of cattle manures in Belgium of 580  $mgkg^{-1}DM$ . In Switzerland, cattle manures generally contain <200 mg Zn/kg DM (Menzi and Kessler, 1998). Webber and Webber 1983) reported cattle manure Zn concentrations in the range of  $30 \pm 225 \text{ mg kg}^{-1} \text{DM}$ . The mean Cu concentration in cattle manures was reported to be  $16 \pm 62 \text{ mg kg}^{-1} \text{DM}$  (Fleming and Mordenti, 1991; Menzi and Kessler, 1998). Sims and Wolf (1994) reported Zn concentrations ranging from non-detectable to 660 mg kg<sup>-1</sup> DM in poultry manure and from non-detectable to 669 mg kg<sup>-1</sup> DM in poultry litter. The maximum content of Cu in poultry manures and litters were 232 mg kg<sup>-1</sup> DM and  $1003 \text{ mg kg}^{-1} \text{ DM}$ , respectively.

Farm manures are a valuable source of major plant nutrients (N, P, K) and organic matter. However, care must be taken to ensure that manure applications containing elevated levels of heavy metals do not cause long-term soil contamination. Then, the typical amounts of heavy metals in solid manures have to be calculated before manure addition to soil. Excessive cumulative application of manure can cause accumulation of phosphate in surface soils, which increases the potential for runoff of phosphate which can cause adverse effects in streams. Thus applications of manures are being limited to the phosphate fertilizer requirement of crops rather than the N fertilizer requirement which applied much higher annual manure amounts. Several groups have developed models for application of major and micronutrients with all soil amendments for large areas, considering livestock manure, biosolids, phosphate and other fertilizers. These studies illustrate accumulation of Zn, Cu and Cd in soils when microelement rich manures are applied (Keller and Schulin, 2003). These issues are more important in areas with intensive animal production because feedstuffs are imported from distant farms, but the livestock residues are applied to the local farm of the livestock producer. In developing countries where livestock production is achieved with local inputs, accumulation of Zn and other elements in the soil is less of a concern.

#### 4.6 Organic Amendments

Biosolids and composts are usually effective Zn and Cu fertilizers. Possibility of correcting micronutrient deficiency in different plant species by application of biosolids, municipal waste leachates, and compost has been reported (Khoshgoftarmanesh and Kalbasi, 2002). When soils need Zn or Cu fertilization, high quality biosolids and composts could be used to supply the required Zn and Cu. Such amendments are not normally able to replace B fertilizers, but mixing with coal combustion byproducts can supply adequate B for the mixture to serve as B fertilizer (Chaney and Ryan, 1993).

Because production of composts will save costs in both urban and agricultural areas, they can be used as inexpensive fertilizers. Although such benefits are possible from use of composts, these products must be safe

for sustainable use in horticulture and agriculture for their use to be permitted by governments, and must reliably supply nutrient and organic matter benefits to become competitive products in the market place. The potential presence of pathogens, heavy metals, xenobiotic compounds, and possible element imbalance in composts has caused concern to compost application. Some believe that because the concentration of Zn or Cu in composts is higher than found in background soils, these materials must not be utilized on soils. However, researchers have used high quality organic matter/compost products for decades without adverse effects (Andersson, 1983; Chaney and Ryan, 1993; Woodbury, 1992). Composts prepared in developing countries often do not receive the contaminants which have caused concerns in developed countries. Over time boron use in glues and other products has declined such that municipal solid wastes which used to commonly contain phytotoxic levels of boron are only a useful B fertilizer today. As with biosolids, composts from off-farm sources should be analyzed for contaminants and nutrients so that these resources can be wisely managed on cropland.

#### 4.7 Crop Rotation

Cropping systems practices influence micronutrient availability. A preceding crop may influence soil conditions that govern soil fertility in general and micronutrients phytoavailability to the successive crop in particular. These can result from residual effects of root litter and exudates on soil physical and chemical parameters as well as on soil microorganisms, generation of root channels, and other plant-soil interactions. Legumes may positively affect soil quality by improving soil physical properties. Improved physical properties of the soil provide a better root growth condition and as a result, enhance uptake of immobile micronutrients by roots. Crop rotation systems may also affect some soil chemical properties (i.e. pH). Alvey et al. (2001) found that a legume rotation system was more effective in increasing the pH of the rhizosphere as compared to continuous sorghum cropping system. The increase in pH was considered as an important mechanism of the plant to cope with acidity-induced nutrient deficiency. However, in the long term, N fixation increases soil acidity

and must be counteracted by alkaline amendments. Khoshgoftarmanesh et al. (2007) reported that wheat plants grown after sunflower accumulated more Zn in their grains as compared to those planted after cotton. Liu et al. (2002) found that a rice-upland crop rotation promoted the reduction of Mn in the surface soil and accelerated the oxidation and accumulation of Mn in the subsoil. Bronick and Lal (2005) reported that crop rotation enhanced soil organic carbon accumulation. Diekow et al. (2005) reported that higher residue input associated with legume-based cropping systems significantly increased soil organic carbon. Mitchell and Entry (1998) found that long-term planting of legumes as a winter cover crop resulted in higher soil organic carbon levels compared with treatments that did not include a winter cover crop.

Wei et al. (2006) found that available Zn and Cu varied with different cropping systems. The amount of phytoavailable Cu in soil decreased in the order: continuous clover > crop–legume rotation > continuous wheat. Differences among cropping systems could probably be attributed to differences in the Cu uptake capacity of each crop. Long-term cultivation of leguminous crops may lead to a large increase in available Mn due to changes in the soil microenvironment by leguminous crops that result in the release of plant available Mn (Williams C.H. and David, 1976).

Gunes et al. (2007) found that intercropping of wheat and chickpea inoculated with a Rhizobium on a calcareous clay loam soil (pH 7.9 in water extract, 0.96% organic matter content) with a very low DTPA-extractable Zn concentration  $(0.11 \text{ mg kg}^{-1})$  increased the Zn concentrations of shoots and seeds in both species. Zuo and Zhang (2008) reported that Fe, Zn, and Cu concentrations of intercropped grown peanut increased significantly compared to those in monocropping both in the greenhouse and field. Systemic mechanisms may be involved in adaptation to nutrient stresses at the whole plant level. A reasonable intercropping system of nutrient efficient species should be considered to prevent or mitigate iron and zinc deficiency of plants in agricultural practice (Zuo and Zhang, 2009).

The effect of crop rotation on the available and total micronutrient contents in soils is somewhat different (Wei et al., 2006). Total nutrient content is affected mainly by crop uptake, and therefore the magnitude of the decrease depends on crop type and uptake intensity.

### 5 Developing Micronutrient-Efficient Genotypes: A New Strategy

### 5.1 Definition of Nutrient Efficiency

According to the Soil Science Society of America (1997), nutrient efficiency is the ability of a plant to absorb, translocate, or utilize more of a specific nutrient than other plants under conditions of relatively low nutrient availability in the soil or growth media. This is the agronomic focus rather than the food quality focus which will be discussed below. Graham (1984) defined plant nutrient efficiency as the ability of a genotype to produce a high yield under conditions in which a soil nutrient is growth-limiting. Blair (1993) defined plant nutrient efficiency as the ability of a plant to acquire nutrient from a growth medium and/or to utilize it in the production of shoot and root biomass or grain. According to Isfan (1993), efficient genotypes are those with high ability to absorb nutrients from soil and fertilizer, produce high grain yield per unit of absorbed nutrient and store relatively little nutrients in the straw. As defined by Clark (1990), an efficient genotype is one that produces more dry matter or has a greater increase in the harvested parts per unit time, area, or applied nutrient, has less deficiency symptoms, or has greater incremental increases and higher concentrations of mineral nutrients than other genotypes grown under similar conditions. Gourley et al. (1994) proposed that the nutrient requirement of an efficient germplasm is less than an inefficient germplasm. Fageria et al. (2008) defined a nutrient efficient genotype as a genotype that produces higher economic yield with a determined quantity of applied or absorbed nutrient than other genotypes under similar growing conditions.

Nutrient efficiency emphasizing utilization is generally defined as total plant biomass produced per unit nutrient absorbed, which is equivalent to the reciprocal of the nutrient concentration of the entire plant (Gourley et al., 1994). This often is called the 'nutrient efficiency ratio' and has been used extensively to describe the internal nutrient requirement, particularly of P, in many agronomic plant species (Baligar et al., 1990; Glass, 1989; Godwin and Blair, 1991). Using this definition, P-efficient alfalfa genotypes with higher shoot P concentration are selected to overcome P nutrition limitations in cattle (Miller et al., 1987). The reciprocal of nutrient concentration does not consider the crop yield. Siddiqi and Glass (1981) suggested "utilization efficiency" defined as the product of yield times the reciprocal of nutrient concentration. The nutrient uptake per unit root length, surface area, or weight has also been defined as "uptake efficiency" (Buso and Bliss, 1988). In most studies, micronutrient efficiency has been calculated as the ratio of yield (shoot dry matter or grain yield) produced under micronutrient deficiency (-M) to yield produced with micronutrient fertilization (+M), as fllows (Graham, 1984):

Micronutrient efficiency =  $(yield - M/yield + M) \times 100$ 

In this paper, the last definition is considered as the main definition of micronutrient-efficiency for separating micronutrient-efficient and inefficient genotypes. Such an index is used to select improved micronutrient-efficient genotypes in a breeding program, or to provide information on qualities of commercial cultivars for growers.

### 5.2 Micronutrient-Efficient Genotypes and Crop Productivity

Application of fertilizers is not a totally successful strategy in alleviating micronutrient deficiency because of agronomic, economic, and environmental (Mortvedt, 1994; Graham and Rengel, 1993; Hacisalihoglu, 2002).

A more efficient and sustainable solution to micronutrient deficiency limitations to crop production is the development and use of micronutrient-efficient plant genotypes that can more effectively grow on soil with low phytoavailable micronutrient contents, which would reduce fertilizer inputs and protect the environment as well. Selection of plant genotypes that can tolerate low nutrient supply may increase productivity on low fertility soils and reduce fertilizer requirements (Gourley et al., 1994).

One of the most important adaptive responses for crop plants involves their ability to deal with soilmediated abiotic stresses involving deficient levels of micronutrients in the soil. Plant species vary significantly in tolerance to micronutrients deficiency stress; some are able to cope with low micronutrients availability, and thus, grow well even when other species or cultivars suffer reduced yield due to micronutrient deficiency (Graham and Rengel, 1993).

Exploiting genetic diversity of plants for enhanced productivity in poor fertility soils is a desirable and important objective in order to meet food demands for an increasing world population. Variation among plant germplasm in the ability to acquire nutrients from the soil has been investigated for decades (Godwin and Blair, 1991). There are several key mechanisms that could be involved in nutrient efficiency (Baligar et al., 2001; Fageria and Barbosa Filho, 2001; Fageria and Baligar, 2003; Khoshgoftarmanesh et al., 2004a, 2006a), including root processes that increase the bioavailability of soil nutrient for root uptake; enhanced root uptake and translocation of nutrients from the root to the shoot; altered subcellular compartmentation of nutrient in shoot cells; and more efficient biochemical utilization of nutrients in cells of the shoot. Many possible processes for control of nutrient efficiency have been investigated. In recent years, more attention has been paid to micronutrient efficiency.

Micronutrient efficiency is genetically controlled and the physiological and molecular mechanisms of micronutrient efficiency of plants are just beginning to be understood. Large ranges of genotypic variation in response to micronutrients deficiency stress have been reported in different plant species, particularly in cereals (Graham et al., 1992; Cakmak et al., 1997a, 1998). Such large variation is promising for developing plant genotypes that are more efficient in root uptake, translocation from root to shoots and/or internal utilization of micronutrients. Micronutrient-efficient genotypes may provide a number of other benefits, such as reductions in the use of fertilizers, improvements in seedling vigor, resistance to pathogens, and enhancement of grain nutritional quality (Graham and Rengel, 1993; Bouis, 1996; Graham and Welch, 1996).

Selecting and breeding staple food crops which are more efficient in the uptake of trace minerals from the soil and load more trace minerals into their seeds combines benefits both for agricultural productivity and human nutrition. This approach may have important applications for increasing farm productivity in developing countries in an environmental-friendly way (Cary et al., 1994; Kannenberg and Falk, 1995). Growing nutrient-efficient genotypes on soils with low nutrient availability would reduce land degradation by reducing the use of machinery (Thongbai et al., 1993) and by minimizing application of chemicals such as fertilizers on agricultural land (Rengel, 2001).

Beneficial effects of micronutrient-efficient crop genotypes for agricultural productivity result in extremely high cost/benefit ratios for investing in this type of micronutrient intervention. Furthermore, the adoption and spread of micronutrient enriched seeds by farmers can be driven by profit incentives because micronutrient enriched seeds increase crop productivity when planted in micronutrient-poor soils (Graham et al., 2001). The benefits can be disseminated widely and they are sustainable once developed unlike current micronutrient interventions that rely on supplementation or food fortification (Graham et al., 2001, 2000; Welch et al., 1997).

Developing plants with a high ability to extract transition metal nutrients from soils may be more useful for Fe and Mn. Soils which cause deficiency contain substantial levels of total Fe or Mn, but the minerals are not phytoavailable to traditional genotypes. Induced iron deficiency chlorosis is widespread and is a major concern for plants growing on calcareous or alkaline soils due to their high pH and low availability of iron (Welch et al., 1991; Marschner, 1995). In calcareous soils, Fe and Mn rapidly convert to immobile forms and generally several foliar applications are required to prevent deficiencies throughout the growing season. Selecting or breeding iron efficient genotypes of corn, soybean, sorghum and rice is considered the best way for correcting Fe deficiency stress on calcareous soils (Graham, 1984; Fageria and Baligar, 2003). Soil application of Cu, Zn, and Mo is more efficient than Mn and Fe fertilization, on most soils, but all transition metal nutrients are not readily translocated within plants on deficient soil (Nable and Webb, 1993; Graham and Webb, 1991).

Several studies have been done to select and breed cultivars with greater abilities to cope with adverse soil conditions. Micronutrient-efficient genotypes were described that have the ability to take up and utilize nutrients more efficiently under abiotic and biotic stresses (Pessarakli, 1999; Alam, 1999; Baligar et al., 2001) or that some micronutrient-efficient genotypes can tolerate abiotic and biotic stresses (Fageria et al., 2008; Khoshgoftarmanesh et al., 2004a, 2006a, b; Marschner, 1995). Plant crop yield under stress could be enhanced by selection or breeding of plants that have high micronutrient efficiency, and ability to interact effectively with

environmental extremes (Fageria et al., 2008). For example, Khoshgoftarmanesh et al. (2004a, b, 2006a, b) and Khoshgoftar and HajiMozaffari (2006) reported that zinc application in a severely calcareous saline soil increased the salt tolerance of wheat genotypes. Zinc-efficient genotypes are reported to be more tolerant to salinity than Zn-inefficient genotypes (Table 1). Bagci et al. (2007) indicated that sensitivity to Zn deficiency stress became more pronounced when wheat plants were drought-stressed. The effect of irrigation on grain yield was maximized when Zn was adequately supplied. Close relationship between water use efficiency and Zn nutritional status of plants has also been suggested (Bagci et al., 2007).

Plant nutrition status may greatly affect predisposition of plants to attack by, or effect of pests and diseases (Fageria et al., 2008). Micronutrient deficiencies reduce the plant tolerance to diseases and insects (Marschner, 1995). The fungicidal effect of manganese (Mn), copper (Cu) and zinc (Zn) are well-known as they have been common constituents of fungicides (Streeter et al., 2001). Copper, B, and Mn are involved in the synthesis of lignin, and phenolic compounds. Silicon plays a role in producing physical barriers against pathogen penetration (Barker and Pilbeam, 2007; Graham and Webb, 1991).

Streeter et al. (2001) reported that Zn-sufficient plants are more tolerant to the effects of root pruning by the fungus than Zn-deficient plants probably due to fungitoxicity of Zn and the role of Zn in the integrity or stability of the host plant's membranes. The increased root exudation by Zn-deficient plants may accelerate the root-rot disease (Thongbai et al., 1993).

**Table 1** Response of five wheat genotypes with different zinc efficiency to salinity and zinc application treatments (Khoshgoftamanesh et al., 2006a)

Shoot dry matter yield (g pot $^{-1}$ )					
NaCl rate (mM)	Dur-3	Kavir	Falat	Rushan	Cross
Without Zn					
0	69.1	82.6	94.1	92.9	99.6
60	59.3	75.4	63.1	85.5	85.3
120	50.2	55.3	61.3	69.5	80.4
180	30.6	49.1	43.5	68.1	58.5
With Zn					
0	80.5	116.5	98.6	93.3	100.8
60	77.1	92.1	77.5	87.2	89.6
120	76.3	84.3	68.2	79.6	88.1
180	35.1	54.4	43.1	71.8	65.7

A relationship between Zn nutrition and severity of some root-rot diseases has been demonstrated in wheat (Thongbai et al., 1993) and medic plants (Streeter et al., 2001). Streeter et al. (2001) suggested that Zn application does not directly inhibit infection by *Rhizoctonia solani*, nor reduce its pathogenicity, but it does strongly increase root growth. Zinc nutrition is beneficial in reducing the occurrence of phyllody virus in white clover (Carr and Stoddart, 1963), take-all in wheat (Brennan, 1992), and charcoal rot development in maize (Pareek and Pareek, 1999).

Silicon applications reduce the severity of fungal diseases such as blast and sheath blight of rice, powdery mildew of barley and wheat, and vermin damage of rice by the plant hopper in the field (Barker and Pilbeam, 2007). Plant genotypes with higher ability to absorb and/or utilize micronutrients seem to be more resistant to disease than micronutrient-inefficient genotypes. The relationship between insect attacks and micronutrient-efficiency of plant genotypes needs further evaluation (Fageria et al., 2008).

Considerable progress has been made in breeding for Zn, Fe, and Mn efficient dicots and monocots and in identifying the genes involved (Marschner, 1995; Rengel, 2001). By comparing efficient and inefficient cultivars, some information on the genetic control of micronutrients efficiency has been obtained (Graham, 1984; Rengel, 2001). This information will accelerate development of more nutrient efficient crop plants by traditional and modern breeding techniques, and provide important information for genetic engineering.

# 5.3 Micronutrient-Efficient Genotypes, Crop Quality and Human Health

#### 5.3.1 Total Concentration of Micronutrients in Plant Based Foods

Different traditional public health interventions have been used to improve micronutrient nutritional status of the target population; i.e. supplementation, food diversification, industrial fortification and biofortification. Supplementation and industrial fortification have been effective in reducing morbidity and mortality resulting from micronutrient deficiencies worldwide, but infrastructure, purchasing power, or access to markets and healthcare systems is required in these approaches, often not available to poor people particularly in rural areas (Mayer et al., 2008).

Supplementation with pharmaceutical micronutrient preparations e.g. vitamin A and Zn capsules can be a rapid and effective way to alleviate micronutrientdeficiency on an individual basis. However, this strategy has generally failed on a population level in developing countries, due to lack of adequate infrastructure and education (Graham et al., 2000; Stein, 2006). For example, vitamin A supplementation programs covered up to 58% of target population over the last decades in 103 priority countries with a goal of preventing blindness caused by vitamin-A deficiency (UNICEF, 2007). Iron supplementation had little effect on the anemia problem in India, because of mismanagement, underfunding, logistic problems, and poor compliance (Vijayaraghavan, 2002).

Food fortification is a strategy that can be applied rather rapidly at the national level without changes in the dietary habits of people. As a public health measure, wheat flour is commonly fortified with iron, margarine with vitamin A and D, and salt with iodine. However, Zn interventions with fortified foods or supplements are still largely experimental. Some countries also fortify wheat flour with zinc. Fortification of maize and wheat flours with Fe has been implemented in Mexico, Indonesia, and South Africa, but this approach still needs further evaluation (Gibson, 2006). Ahmed et al. (2008) concluded that fortification of whole wheat flour is feasible to alleviate Fe and Zn deficiency in vulnerable groups. A disadvantage of food fortification is that, like supplementation, its successful implementation into society requires safe delivery systems, stable policies, appropriate social infrastructures and continued financial support (White and Broadley, 2005; Gibson, 2006).

In the case that foods differ widely in available contents of micronutrients, dietary modification or diversification would seem a straightforward and sustainable way to combat micronutrients deficiency. But changes in dietary habits require individual and societal acceptance, as well as the availability of alternative foods at affordable prices. Poverty is another challenge that makes this strategy less applicable for poor countries.

Due to problems such as inadequate logistics, insufficient compliance, political instability and instability of funding, many of these interventions have failed. In particular, many of these programs never reached the majority of low-income women, infants and children in developing countries (Welch and Graham, 1999).

Using bioavailable micronutrient-dense staple crop cultivars is another approach that could be used to improve the micronutrient nutritional status of human (Bouis, 1996; Combs et al., 1996; Welch and Graham, 1999; Frossard et al., 2000; Welch, 2002). Cultivation of micronutrient-efficient crop genotypes holds great promise for making a significant, low-cost, and sustainable contribution to reducing micronutrient, particularly mineral deficiencies in humans.

This strategy, for which the term "biofortification" has been coined, has the potential to be sustainable as well as economical. Biofortification is based on the principle that health comes from the farm, not the pharmacy (Mayer et al., 2008). Like fortification in general, it does not require people to change their dietary habits. This approach is attractive for the farmer because of increased yields and reduced seedling losses and does not require special infrastructure. In addition, the output of this program can be distributed to the remote rural areas where poor populations are living at low cost (Graham et al., 2000; Yang et al., 2007).

Biofortification can be achieved (i) by breeding or genetic engineering crop plants with improved ability to accumulate target micronutrients in potentially bioavailable form in edible plant parts (Graham et al., 2000; Welch and Graham, 2004; White and Broadley, 2005), (ii) by increasing (bioavailable) micronutrient density in food plants by agricultural methods of crop cultivation, for example by fertilizer application or by amendments that increase the element's bioavailability in the soil for plant uptake (Frossard et al., 2000; Graham et al., 2001; Welch, 2002).

Most modern cultivars of wheat and rice have a lower concentration of micronutrients in grain than traditional cultivars because breeders generally focused on increasing yield, with no attention to the micronutrient concentrations in grain. Breeding micronutrient-dense cultivars of staple foods is a powerful tool to combat micronutrient malnutrition. Recent findings show that it is possible to breed for enhanced levels of iron, zinc and provitamin A carotenoids in edible tissues of rice, wheat, maize, beans and cassava (Graham et al., 2001). Recently, the international genomes of rice (*Oryza sativa* L.), wheat (*Triticum aestivum* L. and *T. durum* Desf.), maize (Zea mays L.), beans (Phaseolus vulgaris L.), cassava (Manihot esculenta Crantz) and sweet potato (Ipomoea batatas, L.) have been surveyed for high micronutrient density traits (Graham et al., 2001).

Large variation in the grain micronutrients concentration e.g. Fe and Zn in different wheat genotypes has been reported (Cakmak, 2002; Graham et al., 1999; Kalayci et al., 1999; Khoshgoftarmanesh et al., 2004a, 2006a, b). As shown in Table 2, different wheat genotypes vary greatly in grain Zn and Fe concentrations, ranging from 8.5 to 84  $\mu$ g g<sup>-1</sup> for Fe and 4.6 to 41.4  $\mu$ g g<sup>-1</sup> for Zn (Khoshgoftarmanesh et al., 2007). The large genetic variation found in Fe, Zn, Mn, and Se concentrations among geotypes in the major germplasm banks is sufficient to justify the possibility of developing micronutrient-efficient genotypes. Thus, it seems feasible for crop breeders to select for high micronutrients density traits in breeding programs.

Modern recombinant DNA technology can also be used to enhance the nutritional quality of food crops such as increasing the amount and bioavailability of micronutrients in plants (DellaPenna, 1999; Frossard et al., 2000; Goto et al., 1999, 2000; Lucca et al., 2001). For example, recombinant DNA technology was used to improve the provitamin A content of rice-grain endosperm in 'Golden rice' (Ye et al., 2000). Goto et al. (1999) transformed rice plants using a phytoferritin (a major protein storage form of Fe in plants) gene from soybean and a rice endosperm. Lucca et al. (2001) also enriched the Fe concentration in ricegrain endosperm using the phytoferritin gene from pea plants.

The strategy of breeding for mineral and vitamin enhancement of staple foods has several

**Table 2** Range and mean concentrations of Fe and Zn in the shoot and grain of different wheat genotypes in the presence and absence of added Zn fertilizer (Khoshgoftarmanesh et al., 2007)

	Concentration ( $\mu g g^{-1}$ )			
	Zn		Fe	
	Minus Zn	Plus Zn	Minus Zn	Plus Zn
Shoot				
Range	5.5-21.3	5.9-26.0	15.0-98.0	14.3-101.0
Mean	9.2	13.5	57.1	59.6
Grain				
Range	4.6-41.4	7.6-36.2	8.5-75.8	9.7-84.1
Mean	9.8	11.8	45.2	61.2

complementary advantages. Micronutrient-dense crop genotypes deploy micronutrients to consumers through the traditional ways with no necessity for changes in consumers' behavior (Mayer et al., 2008). The consistent daily consumption of large amounts of food staples by all family members is another benefit of this approach. Nevertheless, any intervention to improve micronutrient status must target women, children, and adolescents because of their elevated needs for minerals and vitamins (Nicklas, 1995).

Biofortification programs for  $\beta$ -carotene, iron, and zinc have been estimated to be cost-effective (Neidecker-Gonzales et al., 2007; Stein et al., 2007). Investments in plant-breeding research are much less than those in supplementation and fortification programs because the increased value persists after development. In addition, supplementation and fortification programs must be sustained and thus, much more investment is needed in these programs. If investments are not sustained, benefits disappear. Development of iron- and zinc-dense cultivars of rice or wheat might cost as much as \$10 million each over 10 years, including the costs of nutrition efficacy tests, the costs of dissemination in selected regions, and the costs of a nutrition and economic impact evaluation. Moreover, benefits are sustainable at low maintenance costs.

However, before beginning extensive breeding programs, it is essential to confirm that the micronutrient is sufficiently well absorbed and utilized by human consumers. In fact, further research is needed to determine if the edible portions of micronutrient dense crop grains still retain enriched levels of micronutrients after milling and processing, and if enriched levels of micronutrients in grain are bioavailable to target human populations.

Agronomic biofortification is required for optimizing and ensuring the success of genetic biofortification of cereal grains with Zn and Se (Broadley et al., 2006; Cakmak, 2008). In case of greater bioavailability of the grain Zn derived from foliar applications than from soil, agronomic biofortification would be a very attractive and useful strategy in solving Zn deficiency-related health problems globally and effectively (Cakmak, 2008). In fact, agronomic biofortification strategy appears to be essential in keeping sufficient amount of available Zn in soil solution and maintaining adequate Zn transport to the seeds during reproductive growth stage (Cakmak, 2008).

#### 5.3.2 Bioavailability of Micronutrients in Food

Although increasing the concentration of micronutrients in plant foods is necessary, the absorption and utilization (i.e., bioavailability) of micronutrients in meals containing plant foods may be also improved (Graham and Welch, 1996). Plant foods contain various substances that interact with micronutrients. Some, called antinutrients, reduce whereas others, called promoters, enhance micronutrient bioavailability to humans (Graham et al., 2001). These substances can be also manipulated by agricultural practices (Graham et al., 2001) and by genetic manipulation (Lucca et al., 2001).

The nutritional value of a diet can not be determined solely from the total contents of individual nutrients. Concentrations of antinutrients and promoters as well as interactions between nutrients affect bioavailability, which is the amount of nutrient absorbed from the diet (Graham et al., 2000). In fact, the fraction of micronutrient content in food that is available for absorption by the human gastro-intestinal tract depends on the presence of other food ingredients. A number of nutrient promoters and antinutrients has been found. Research on the potential role of a variety of substances to act as promoters or antinutrients of some micronutrients such as Fe and Zn bioavailability to humans has been reviewed by House (1999). There is evidence for a promoter function of certain chelating low-molecularweight organic acids as well as amino acids. Drakakaki et al. (2005) concluded that the expression of recombinant ferritin and phytase could help to increase iron availability and enhance the absorption of iron, particularly in cereal-based diets that lack other nutritional components. Phytate, tannins and other polyphenols, and some heavy metals act as antinutrients (Frossard et al., 2000; Graham et al., 2001).

Plant crop genotypes differ not only in grain micronutrient concentrations but also in the amounts of antinutrients and promoters in their seeds. For example, phytate concentration strongly depends on the plant genotype (Cakmak and Marschner, 1986; Khoshgoftarmanesh et al., 2007; Marschner, 1995, Torun et al., 2001). Erdal et al. (2002) found differences in phytate content of grain in 20 different wheat cultivars grown on a Zn-deficient calcareous soil (0.10 mg kg<sup>-1</sup> DTPA-extractable Zn) in central Anatolia. Zinc fertilization (23 kg ha<sup>-1</sup> Zn applied as ZnSO<sub>4</sub>) increased seed Zn concentration in all cultivars and

**Table 3** The range and mean molar ratio of phytic acid to Zn in grains of selected wheat genotypes in the absence and presence of added Zn (Khoshgoftarmanesh et al., 2007)

	· · ·		. ,		
	Phytic acid/Zn molar ratio				
	Whole grain		Bran-less flour		
	Minus Zn	Plus Zn	Minus Zn	Plus Zn	
Range	39–88	43–76	27–49	18–37	
Mean	48	31	29	26	

reduced seed phytate in the majority of the cultivars. The average increase in seed Zn was from 8.8 to 16.7 mg kg<sup>-1</sup>, while phytate decreased in average from 10.7 to 10.0 mg g<sup>-1</sup> grain dry weight. Raboy et al. (1991) showed the existence of an important variation in phytate-P among four wheat cultivars and their 60 F6 lines. Barrier-Guillot et al. (1996) reported that phytate-P in seeds of four wheat cultivars varied between 0.92 and 2.80  $g kg^{-1}$  DW. The range and mean concentration of Zn and molar ratio of phytate-to-Zn in selected Iranian wheat genotypes are shown in Table 3. The genotypic variability in seed phytate concentration of wheat genotypes indicates that this variation can be exploited for breeding genotypes with low phytate concentrations to improve Zn bioavailability in cereal-based foods although attention should be paid to the fact that low phytate content increases the possibility of cancers. In addition, phytate is required for seed germination.

In most cases, phytate-to-Zn molar ratios in foods are considered a predictor of Zn bioavailability (Gargari et al., 2007). This index has been used widely (Gibson et al., 2003; International Zinc Nutrition Consultative Group-IZiNCG, 2004), and it is considered as a good index for zinc bioavailability by the World Health Organization (World Health Organization, 1996) and International Zinc Nutrition Consultative Group (International Zinc Nutrition Consultative Group (International Zinc Nutrition Consultative Group-IZiNCG, 2004).

## 5.4 Combining the Benefits of High Yield and Better Nutritional Quality

An important question in development of micronutrient-efficient genotypes is the possibility of combining high yield with better micronutrient nutritional quality. Previous studies showed that it is possible to combine micronutrient-rich traits with high yield. Both seedling vigor and nutritional quality can be improved through genetically modifying seeds with micronutrient enrichment traits. The highest micronutrient densities, which are approximately twice as high as those popular modern cultivars and indicating the existing genetic potential, can be successfully combined with high yield.

The combining of benefits for human nutrition and agricultural productivity, resulting from breeding staple food crops which are more efficient in the uptake of trace minerals from the soil and which load more trace minerals into their seeds, results in extremely high benefit-cost ratios for investments in agricultural research in this area. This approach would be more valuable and cost-effective by estimating the costs paid to heal micronutrient malnutrition in developing countries. Some adverse effects of Zn and Fe deficiency cannot be remediated by supplying adequate levels of Zn or Fe later in childhood, so prevention needs to be the focus of this seed improvement program.

High trace mineral density in seeds produces more viable and vigorous seedlings in the next generation, and efficiency in the uptake of trace minerals improves disease resistance, agronomic characteristics which improve plant nutrition and productivity in micronutrient deficient soils (Welch, 1999; Yilmaz et al., 1998). Adoption and spread of nutritionally-improved varieties by farmers can rely on profit incentives, either because of agronomic advantages on trace mineral deficient soils or incorporation of nutritional improvements in the most profitable varieties being released (Harris et al., 2008).

It has been shown that wheat plants grown from seed with high Zn content can achieve higher grain yields than those grown from the low-Zn seed when Zn was not applied to the soil (Yilmaz et al., 1998). Therefore, sowing seeds with higher Zn contents can be considered a practical solution to alleviate plant Zn deficiency especially under rainfed conditions, in spite of it being insufficient to completely overcome the problem (Yilmaz et al., 1998).

Mineral-packed seeds sell themselves to farmers because these trace minerals are essential in helping plants resist disease. More seedlings survive and initial growth is more rapid. Ultimately, yields are higher, particularly on trace mineral "deficient" soils in arid regions. Because roots extend more deeply into the soil and so can tap more subsoil moisture and nutrients, the mineral-efficient varieties are more drought-resistant and so require less irrigation. And because of their more efficient uptake of existing trace minerals, these varieties require lesser chemical inputs. Thus, the new seeds can be expected to be environmentally beneficial as well. It is conceivable that seed priming by spraying seed fields with Fe and Zn fertilizers during grain filling will provide enough additional yield benefit to justify additional seed price. But this approach is insufficient to improve the density of bioavailable micronutrients in grain of crops grown with "primed" seed. Several different commercial practices may be beneficial to improve plant production despite low levels of soil micronutrients in many nations.

### 5.5 Need to Limit Grain Cd and As While Improving Levels of Zn and Fe in Crops

In contrast with the essential nutrients Zn and Fe, cadmium (Cd) in soils can be accumulated to levels in rice and some other crops that are dangerous to consumers (Chaney et al., 2004). In areas of Asia where mine wastes contaminated rice paddy soils, rice accumulated Cd to levels over  $0.4 \text{ mg kg}^{-1}$ , the CODEX ALIMENTARIUS maximum recognized level safe for lifetime consumption by humans. In Japan, China and Thailand, mine-contaminated paddy soils have caused adverse effects in exposed subsistence farm families (Nogawa et al., 2004; Cai et al., 1998; Swaddiwudhipong et al., 2007). The nature of rice production and composition of polished rice grain contribute to the potential for this effect: (1) when flooded rice is drained at flowering, rapid drop in soil pH and transformation of CdS to more soluble forms allows rapid uptake of Cd to rice grain during grain filling; (2) polishing of rice removes much of the Zn, Fe and Ca in brown rice; levels present in polished grain are insufficient to support human health; and (3) subsistence farming individuals may consume home-grown rice for their lifetime, allowing accumulation in the kidney over time until injury occurs. Although other crops require consideration regarding transfer of soil Cd to humans, western populations have not been found to suffer Cd diseases, perhaps because few consume home-grown staple foodstuffs grown in contaminated soils for their lifetime. Further, most soil Cd is accompanied by 200-fold higher Zn, so higher Zn tends to limit Cd uptake into edible crop tissues (Chaney et al., 2004).

Two aspects of breeding Fe and Zn improved crops should consider the potential for Cd accumulation. First, cultivars vary in Cd accumulation due to genetic variation (Grant et al., 2008); normal breeding practices can produce higher Cd progenies by chance alone. In general, Cd has been found to enter plants on the root Zn-transporters (Hart et al., 2002). Thus breeding to increase Zn uptake may cause increased Cd accumulation as well. Transport to grain of Cd and Zn are somewhat independent, so it is possible to achieve both higher Zn and Cd uptake but only higher Zn movement to grain. Breeding programs to improve grain Zn and Fe should check to assure that Cd is not being increased along with Zn. In the same way, breeding to increase aluminum tolerance to allow production in strongly acidic soils will tend to increase uptake of Cd and other divalent cations (McLaughlin et al., 1999). Breeders of acid tolerant crops need to assure that their new cultivars are not also increased in Cd.

Arsenic (As) is usually independent of grain Fe and Zn, but recent issues of As in irrigation water and rice require similar effort to consider the potential for As accumulation in grain. Because rice is grown in flooded soils, and flooding produces arsenite which is more soluble, more phytotoxic, and more accumulated by rice, using high As waters to produce rice may produce grain with excessive As for lifetime consumption by humans (Meharg, 2004; Abedin et al., 2002). For rice cultivars to be grown on such As enriched soils, breeding lower As accumulating genotypes may be critical to the continued success of agriculture (Williams et al., 2005).

# 5.6 Challenges in Programs for Selecting or Breeding Micronutrient-Efficient Genotypes

Many different definitions for efficiency make the use of the term ambiguous. These differences in defining nutrient efficiency makes it difficult to compare results of different studies particularly by considering that nutrient efficiency is influenced by other factors, such as soil fertility status, climate, crop rotations, and management practices (Stewart et al., 2005). Gerloff and Gabelman (1983) proposed that genotypes differing in yield under nutrient stress should only be designated efficient or inefficient if they are normal in appearance and have similar yields when an optimal amount of the nutrient is available. To reduce the confounding effects that other factors have on nutrient efficiency, Gourley et al. (1994) proposed that equivalent yields of genotypes be demonstrated where nutrients are not limiting. This advise has not been widely followed by researchers; plant species and cultivars with significantly different genetic potentials have been compared for their nutrient efficiency. Figure 1 shows models of the response of three plant varieties with different potential yield to different soil nutrient levels. These plant varieties have different yields in the presence of an optimal amount of the nutrient. The plant variety 3 produced the highest yield under non-stress conditions while its nutrient efficiency, determined as commonly used efficiency definition, is less than other varieties. In contrast, the most nutrient-efficient variety 1 had the lowest yield when an optimal amount of the nutrient is available.

One of the major difficulties with breeding for resistance to nutrient deficiency is that the stress condition is frequently assessed based only on plant growth. However, growth is as a very complex character influenced by different inter-related processes. Therefore,





**Fig. 1** Suggestive response of three different plant varieties to nutrient concentration in media. Genotype 3 produced the highest yield under non-stress conditions while its nutrient efficiency, determined as commonly used efficiency definition, is less than other genotypes. In contrast, the most nutrient-efficient Genotype 1 had the lowest yield when an optimal amount of the nutrient is available

growth is not a suitable parameter for distinguishing causes and effects of micronutrient deficiency stress.

Better screening techniques need to be developed by targeting specific processes that promote micronutrient deficiency, rather than those that appear as a consequence of it. It is expected that in the near future molecular methods will dominate selection for nutrient efficiency. Development of molecular markers for regions of the genome segregating with the trait of a particular nutrient efficiency will be crucial in that respect. However, molecular markers may be populationdependent, making them unsuitable for use in breeding programs. Even in screening based on molecular methods, seed nutrient content may need to be measured to assure the effectiveness of the program (Rengel, 2001).

Identification of genotypes with differing nutrient efficiencies, by whatever definition, generally includes investigation of potential morphological, physiological, and biochemical mechanisms involved. These mechanisms have been well reviewed (Caradus, 1990; Sauerbeck and Helal, 1990). However, it is often difficult to separate cause from effect when evaluating potential mechanisms of efficient nutrient uptake and utilization. Mechanisms conferring micronutrient efficiency are diverse and complex. Considering the close relationship between the root and shoot activities, differences in yield or nutrient accumulation by plants may incorrectly be attributed to differences in root morphology and function (Gourley et al., 1994).

Another challenge in breeding programs for assessment of a large number of genotypes is that field testings are time-consuming and expensive. Therefore, simple, fast and inexpensive techniques rather than expensive field experiments are required (Graham, 1984; Graham and Rengel, 1993). At this date, valid field screening is still needed before improved cultivars can be identified. Although genes may be identified in laboratory and greenhouse programs, incorporating micronutrient efficiency or density in grain with high yield and other agronomic performance measures will require field screening using traditional breeding approaches.

In most cases the crop yield has been considered as the main basis for separating micronutrient-efficient and inefficient genotypes (Clark, 1990; Stewart et al., 2005; Fageria et al., 2008). There is a concern whether or not micronutrient-efficient genotypes with higher tolerance to micronutrient deficiency can accumulate more nutrients in their edible parts. Although micronutrient-efficient genotypes possess higher uptake capacity, they do not necessarily have higher micronutrient concentrations in their leaves or grains (Graham et al., 1992). Micronutrient-efficient genotypes may even contain lower micronutrient concentrations in their edible parts than micronutrientinefficient genotypes (Cakmak et al., 1997a, b, 1998). Enhanced micronutrient uptake by efficient genotypes under micronutrient deficiency stress improves dry matter production and often results in decreased tissue micronutrient concentration (dilution effect) to levels similar to those present in micronutrient-inefficient genotypes (Marschner, 1995). For example, Cakmak et al. (1997a) reported that in Zn-deficient soil, the most Zn-efficient rye had lower Zn concentration in its tissues than a Zn-inefficient durum wheat. Khoshgoftar et al. (2006a) reported that Zn-efficient wheat genotypes had lower shoot and grain Zn concentrations as compared to Zn-inefficient genotypes although the total amount of Zn per shoot or grain was higher in Znefficient genotypes due to more shoot dry matter yield.

Two important questions need to be answered in programs for selecting or breeding micronutrientefficient cultivars. The first question is whether there is stability in response to nutrient efficiency of plant genotypes in different environments or how do the interactions of genotype by environment (G×E) affect response of genotypes to nutrient deficiency and fertilization condition? The second question regards relationships between crop productivity, mineral concentrations and other nutritional quality factors (e.g. concentrations of promoters and antinutrients) and the way that both crop productivity and grain quality aspects can be combined in developing new micronutrient-efficient genotypes.

Another important aspect that has to be considered in developing micronutrient-dense genotypes is that the bioavailability of micronutrients in enriched genotypes must be tested in humans to assure that they are of benefit to people preparing and eating them in traditional ways within normal household environments. Consumer acceptance must be tested (taste and cooking quality must be acceptable to household members) to assure maximum benefit to nutritional health.

# 5.7 Stress Tolerance Indicators for Selecting Micronutrient-Efficient Genotypes

Micronutrient deficiencies are abiotic stresses for plant production and most researchers have considered micronutrient efficiency synonymous with plant tolerance to micronutrient deficiency stress. A better understanding of the stress tolerance indicators is required for developing more reliable screening procedures to identify and select genotypes with high micronutrient efficiency. Breeders have used several yield stability analyses for identifying tolerant and sensitive genotypes to different environmental stresses (Fernandez et al., 1989). A relatively severe deficiency of an essential nutrient can be considered as a stress for plants (Fernandez, 1991). The field stress environment is characterized primarily by low inputs, suboptimal levels of irrigation, nutrients, temperature, and plant protection measures (Blum, 1988). Selection of genotypes adapted to both stress and non-stress environments have been the main objective of several yield trials.

Selection criteria are proposed to select genotypes based on their performance in stress and non-stress environments (Rosielle and Hamblin, 1981) who defined stress tolerance (TOL) as the difference in crop yield between the stress ( $Y_s$ ) and non-stress environment ( $Y_p$ ), and mean productivity (MP) as the average yield of  $Y_s$  and  $Y_p$ . Fischer and Maurer (1978) proposed a stress susceptibility index (SSI), expressed by the following relationship:

$$SSI = [1 - (Y_s/Y_p)]/SI$$
 (1)

where SI is the stress intensity and is estimated as  $[1 - (Y_s/Y_p)]$ , and where  $Y_{s-}$  and  $Y_{p-}$  are the mean yields over all genotypes evaluated under stress and non-stress conditions. Rosielle and Hamblin (1981) evaluated the genotypic selection based on mean productivity (MP) and stress tolerance (TOL) and showed that selection of genotypes based on stress tolerance was efficient in improving yield under stress conditions, whereas the selected genotypes performed poorly under non-stress environments.

The heritability for yield is sometimes higher in the non-stress environment compared with the stress environment (Frey, 1964). Generally, evaluation in the non-stress environment allowed a better expression of genotypic potential, with higher heritability estimate yield and yield components than genotypes evaluated under the stress environments. Genotype effect and its interaction with environment are usually higher under favorable conditions compared to stress environments because the non-stress environmental conditions allow the genotypes to express their maximum genetic potential.

Genotypes can be categorized based on their performance in stress and non-stress environments into four groups: (A) genotypes that are not affected by stress and produce the same in both stress and nonstress environments; (B) genotypes that produce high yield only in non-stress environments; (C) genotypes with high yield in stress condition and low yield in non-stress environment; (D) and genotypes with low yield in both stress and non-stress environments. The optimum criterion should be separation of Group A from the other three groups. However, the stress tolerance indicators, mean productivity, stress tolerance, and SSI, failed to distinguish Group A genotypes from the other three groups. Fernandez (1991) presented a new stress tolerance index, STI, which can be used to identify genotypes that produce high yields under both non-stress and stress environments.

The stress intensity (SI) is defined as:

$$SI = 1 - (Y_{s-}/Y_{p-})$$
 (2)

where  $Y_{p-}$  is the potential yield of a given genotype in a non-stress environment and  $Y_{s-}$  is the yield of a given genotype in a stress environment. The stress index can take values between 0 and 1. The larger value of SI indicates more severe stress conditions. Mean productivity (Eq. (3)) favors higher yield potential and lower stress tolerance. Rosielle and Hamblin (1981) showed that under most yield trials, the correlations between mean productivity and  $Y_p$ , as well as between mean productivity and  $Y_s$ , would be positive. Thus, selections based on mean productivity generally increase the average yield in both stress and non-stress environments. However, mean productivity can not distinguish between Group A and Group B genotypes.

$$MP = (Y_s + Y_p)/2 \tag{3}$$

The tolerance index (TOL) is calculated by equation (4). A larger value of the tolerance index
represents relatively more sensitivity to stress, thus a smaller value of tolerance index is favored. Selection based on tolerance index separates genotypes with low yield potential under non-stress conditions and high yield under stress conditions. The tolerance index fails to distinguish between Group C and Group A.

$$TOL = Y_p - Y_s \tag{4}$$

The stress susceptibility index (SSI) is another indicator to screen genotypes. Smaller values of SSI indicate greater tolerance to stress. Under most yield trials tolerance index and stress susceptibility index are positively correlated. Selection based on stress susceptibility index (SSI) favors genotypes with low yield under non-stress and high yield under stress conditions. Thus, SSI also fails to distinguish Group A from Group C.

Geometric mean productivity (GMP) is a better indicator than MP to separate Group A and Group C.

$$GMP = \sqrt{(Y_s * Y_p)} \tag{5}$$

Fernandez (1991) proposed stress tolerance index (STI) given by the following equation

$$STI = (Y_p)(Y_s)/(Y_{p-})^2.$$
 (6)

A high value of STI indicates great tolerance to stress and high potential yield. The stress intensity value is also incorporated in the estimation of STI. Thus, STI is expected to distinguish Group A from Group B and Group C.

Three-dimensional plots of  $Y_s$  (x-axis),  $Y_p$  (y-axis) and STI (z-axis) based on grain yield of different bread wheat genotypes grown under severely Zn-deficient soil (stress condition) and Zn fertilized soils (nonstress condition) is shown in Fig. 2. The method presented by Fernandez (1991) was used to show interrelationships among these three variables, to separate the Group A genotypes from the other groups (Groups B, C, D). To compare with the micronutrient efficiency index commonly used by plant nutritionists, a correlation was made between the STI and Zn efficiency of wheat genotypes. It seems that STI is a good selection criterion for identifying high-yield and stresstolerant genotypes. The X-Y plane is divided into four regions by drawing intersecting lines through Y<sub>s</sub> and Y<sub>p</sub> and the four groups are marked as Group A to

Group D (Fig. 2). Most of the Group A genotypes showed high STI and genotypes in group D showed low STI values. Analysis of variance for grain yield and stress tolerance indices for wheat grown in zinc stress and non stress conditions indicates that the STI indicator is less affected by the environment and its stability is more than other indices (Table 4). Table 5 presents the calculated stress tolerance indices and mean grain yield of studied wheat genotypes grown in zinc stress and non stress conditions. Correlation coefficient between stress tolerance indices and yield in zinc stress and non stress conditions is also shown in Table 6. The values of STI are positively correlated to the grain yield produced both in the presence and absence of added Zn. In addition, a very high and positive correlation was found between the STI and MP indicator while there was no significant relationship between the STI values and the amounts of Zn-efficiency calculated by the common formula (Ys/Yp) used by several researchers. It has been previously mentioned that some Zn-efficient genotypes have lower potential yields (Kalayci et al., 1999; Khoshgoftarmanesh et al., 2004a).

The same results were achieved where a threedimensional plot among  $Y_s$  (x-axis),  $Y_p$  (y-axis) and STI (z-axis) was drawn based on grain yield of different bread wheat genotypes grown in severely Fedeficient (stress condition) and Fe-fertilized soils (nonstress condition) (Fig. 3). Higher values of STI were related to Genotypes in Group A (genotypes with higher Fe efficiency and greater potential yield both in the presence and absence of added Fe).

It seems using these stress indicators for separating proper micronutrient-efficient genotypes with higher potential yields would be useful in future studies. Additionally, developing new stress indicators through including other desirable criteria (i.e. concentrations of micronutrients as well as antinutrients and promoters in grains) is very useful in modern breeding programs. In this way, it would be possible to divide crop genotypes into different groups based on both qualitative and quantitative aspects.

### 6 Conclusion

We reviewed different possible approaches including soil and foliar fertilization, crop systems, application



**Fig. 2** The 3-D plot among salt tolerance index (STI), -Ze and +Ze condition for different winter wheat cultivars. Group A, genotypes that are not affected by stress and produce the same in both stress and non-stress environments; Group B, genotypes that produce higher yield only in non-stress environments; Group C, genotypes with higher yield in stress condition and lower yield in non-stress environment; and Group

D, genotypes with low yield in both stress and non-stress environments. Mahdavi (1), Bezostaya (2), Navid (3), Alamoot (4), Alvand (5), Zarin (6), MV-17 (7), Gaspard (8), Gascogen (9), Sayson (10), Shahriar (11), Toos (12), Sabalan (13), Roshan Winter Back Cross (14), Shahi Cross (15), Shahpasand (16), Omid (17), Azar2 (18), Sardari (19) and C-81-10 (20) (Khoshgoftarmanesh, not published)

**Table 4** Analysis of variance for grain yield and stress tolerance indices in wheat in zinc stress and non stress conditions. This table shows the effect of environment on different stress indices and their relative stability. GMP, STI, SSI, MP, TOL, EF,  $Y_{p}$ ,

and Y<sub>s</sub> are geometric mean productivity, stress tolerance index, stress susceptibility index, mean productivity, tolerance index, zinc-efficiency, grain yield under with added Zn treatment, and grain yield under without added Zn treatment, respectively

					r U				. 1 .
GMP	STI	SSI	MP	TOL	EF	Y <sub>p</sub>	Ys	df	Source of variation
Pr > F									
<.0001	0.4948	0.7774	<.0001	0.9666	0.7154	<.0001	<.0001	1	Place
<.0001	<.0001	0.4551	<.0001	0.3119	0.3153	<.0001	<.0001	19	Genotype

of soil amendments and organic sources, and genetic and agronomic biofortification to correct micronutrients deficiency and to increase their density in edible parts of plants. Soil application of fertilizer is the easiest way to correct micronutrient deficiency; however, it is not always successful and sustainable due to agronomic and economic factors such as reduced availability of mi-

**Table 5** Means of yield and stress tolerance indices in zinc stress and non stress conditions in wheat. GMP, STI, SSI, MP, TOL, EF,  $Y_p$ , and  $Y_s$  are geometric mean productivity, stress tolerance index, stress susceptibility index, mean productivity, tolerance index.

erance index, zinc-efficiency, grain yield under with added Zn treatment, and grain yield under without added Zn treatment, respectively

Cultivar	Yp	Ys	EF	TOL	MP	SSI	STI	GMP
Mahdavi	7112	7664	110	-552	7388	-27.9	1.16	7352
Bezostaya	5957	5812	99.5	146	5884	-21.8	0.734	5868
Navid	7928	7222	92	706	7575	119	1.20	7533
Alamoot	7566	7963	106	-397	7764	58.6	1.30	7750
Alvand	8389	8575	102	-186	8482	43	1.53	8470
Zarin	6973	7580	109	-607	7277	90	1.13	7250
MV-17	6647	6293	95.9	354	6470	-135	0.900	6453
Gaspard	8397	7332	88	1034	7850	-75.6	1.35	7814
Gascogen	6611	7131	108	-520	6871	-12.6	0.99	6853
Sayson	7703	7617	99.2	86.2	7660	-50.5	1.27	7646
Shahriar	7585	8029	107	-444	7807	-55.0	1.31	7786
Toos	7712	8054	108	-343	7883	66	1.34	7845
Sabalan	5500	5564	106	-63	5532	-60	0.651	5497
Roshan W.B.C	7384	7552	102	-168	7468	63	1.19	7454
Shahi Cross	6861	6410	94	451	6636	43.3	0.91	6580
Shahpasand	4838	3930	86	907	4384	30.5	0.392	4303
Omid	4706	4673	106	32.7	4689	159	0.464	4653
Azar2	6285	5182	90	1102	5734	-2.9	0.674	5646
Sardari	5710	4628	84	1083	5169	5.1	0.558	5107
C-81-10	7149	7236	103	-87	7192	-128	1.12	7163
LSD 5%	1272	812	21.5	1505	757	217	0.21	747

**Table 6** Coefficient of correlation between stress tolerance indices and yield in zinc stress and non stress conditions in wheat. GMP, STI, SSI, MP, TOL, EF,  $Y_p$ , and  $Y_s$  are geometric mean productivity, stress tolerance index, stress susceptibility

index, mean productivity, tolerance index, zinc-efficiency, grain yield under with added Zn treatment, and grain yield under without added Zn treatment, respectively

mean pre	near productivity, success toterance index, success susceptionity							
Trait	Y <sub>s</sub>	Y <sub>p</sub>	EF	TOL	MP	SSI	STI	GMP
Ys	1							
Yp	0.91**	1						
EF	0.16 <sup>ns</sup>	0.55*	1					
MP	$-0.24^{ns}$	-0.63**	$-0.97^{**}$	1				
TOL	0.97**	0.98**	0.38 <sup>ns</sup>	$-0.47^{*}$	1			
SSI	$-0.12^{ns}$	$-0.06^{ns}$	$0.09^{ns}$	$-0.07^{ns}$	$-0.08^{ns}$	1		
STI	0.96**	0.97**	0.38 <sup>ns</sup>	$-0.46^{*}$	0.99**	$-0.07^{ns}$	1	
GMP	0.97**	0.98**	0.39 <sup>ns</sup>	$-0.47^{*}$	1.00**	$-0.08^{ns}$	0.99**	1

\* and \*\*: significant at the 5% and 1% level of probability, respectively. ns: non significant.

cronutrient element due to topsoil drying, subsoil constraints, disease interactions, and cost of fertilizer in developing countries. Foliar fertilization is more effective than soil application, but too expensive for resource-poor farmers. Seed priming is also an effective way to increase yield and is important for resourcepoor farmers. It seems a reasonable intercropping or rotation system of nutrient efficient species will help to prevent or mitigate micronutrient deficiency of plants in agricultural practice. Selecting and breeding staple food crops which are more efficient in the uptake of trace minerals from the soil and which load more trace minerals into their seeds combines benefits both for agricultural productivity and human nutrition. Plant crop yield under stress could be enhanced by selection or breeding of plants that have high micronutrient efficiency, and ability to interact effectively with environmental extremes. Cultivation of micronutrient-efficient crop genotypes may have important applications for in-



**Fig. 3** The 3-D plot among stress tolerance index (STI), -Fe and +Fe condition for different winter wheat cultivars. Group A, genotypes that are not affected by stress and produce the same in both stress and non-stress environments; Group B, genotypes that produce higher yield only in non-stress environments; Group C, genotypes with higher yield in stress condition and lower yield in non-stress environment; and Group D,

creasing farm productivity in developing countries in an environmental-friendly way. This approach also holds great promise for making a significant, low-cost, and sustainable contribution to reducing micronutrient deficiencies in humans.

The combining of benefits for human nutrition and agricultural productivity, resulting from breeding improved staple food crops results in extremely high benefit-cost ratios for investments in agricultural research in this area. This approach would be more valuable and cost-effective by estimating the costs paid to heal micronutrient malnutrition in developing countries.

genotypes with low yield in both stress and non-stress environments. Mahdavi (1), Bezostaya (2), Navid (3), Alamoot (4), Alvand (5), Zarin (6), MV-17 (7), Gaspard (8), Gascogen (9), Sayson (10), Shahriar (11), Toos (12), Sabalan (13), Roshan Winter Back Cross (14), Shahi Cross (15), Shahpasand (16), Omid (17), Azar2 (18), Sardari (19) and C-81-10 (20) (Khoshgoftarmanesh et al., 2007)

Despite beneficial effects of biofortified genotypes, we found the following major challenges in programs for selecting or breeding micronutrient-efficient genotypes: (1) The stability in response to nutrient efficiency in different environments and relationships between crop productivity, mineral concentrations, and other nutritional quality factors (e.g. concentrations of promoters and antinutrients) has less been considered. (2) There is no reliable indicator for identifying micronutrient-efficient genotypes with high grain yield. It seems that the stress tolerance index, STI, can be used to identify genotypes that produce high yields under both non-stress and stress environments. (3) In most cases, the crop yield has been considered as the main basis for separating micronutrient-efficient and inefficient genotypes while micronutrient-efficient genotypes may even contain lower micronutrient concentrations in their edible parts than micronutrient-inefficient genotypes. Therefore, combining crop productivity and grain quality aspects such as total content and bioavailability of nutrients in developing new micronutrient-efficient genotypes is necessary. (4) Breeding programs to improve grain micronutrient concentration should check to assure that a toxic metal such as Cd is not being increased along with micronutrients.

### References

- Abedin M.J., Cotter-Howells J., Meharg A.A. (2002) Arsenic uptake and accumulation in rice (*Oryza sativa* L.) irrigated with contaminated water, Plant Soil 240, 311–319.
- ACC/SCN (1992) Second Report on the World Nutrition Situation, Vol. I, Global and Regional Results, in: Garcia M., Mason J. (Eds.), pp. 1–80. ACC/SCN, United Nations, Geneva, Switzerland.
- ACC/SCN (1997) United Nations Report on World Nutrition. The 3rd Report. Chapter 2: Micronutrients, pp. 19–52, ACC/SCN, United Nations, Geneva, Switzerland.
- ACC/SCN (2000) Ending malnutrition by 2020: An agenda for change in the millennium. Commission Report, pp. 1–110, ACC/SCN, United Nations, Geneva, Switzerland.
- Afyuni M., Khoshgoftarmanesh A.H., Dorostkar V., Moshiri R. (2007) Zinc and Cadmium content in fertilizers commonly used in Iran. International Conference of Zinc-Crops, May 24–28, Istanbul, Turkey.
- Ahmed A., Anjum F.M., Rehman S.Ur., Randhava M.A., Farooq U. (2008) Bioavailability of calcium, iron and zinc fortified whole wheat flour Chapatti, Plant Food. Hum. Nutr. 63, 7–13.
- Alam S.M. (1999) Nutrient uptake by plants under stress conditions, in: Pessarakli M. (Ed.), Handbook of plant and crops stress, Marcel Dekker, New York, pp. 285–313.
- Alexandratos N. (1995) World Agriculture Towards 2010: An FAO Study, John Wiley and Sons, Chichester, West Sussex.
- Allen L.H. (2000) Ending Hidden Hunger: the History of Micronutrient Deficiency Control. Background Analysis for the World Bank-UNICEF Nutrition Assessment Project World Bank.
- Alvarez J.M., Gonzalez D. (2006) Zinc transformations in neutral soil and zinc efficiency in maize fertilization, J. Agr. Food Chem. 54, 9488–9495.
- Alvey S., Bagayoko M., Neumann G., Buerkert A. (2001) Cereal/legume rotations affect chemical properties and biological activities in two West African soils, Plant Soil 231, 45–54.
- Andersson A. (1983) Composted municipal refuse as fertilizer and soil conditioner. Effects on the contents of heavy met-

als in soil and plant, as compared to sewage sludge, manure and commercial fertilizers, in: Berglund S., Davis R.D., L'Hermite P. (Eds.), Utilization of Sewage Sludge on Land: Rates of Application and Long-Term Effects of Metals, D. Reidel Publ., Dordrecht, pp. 146–156.

- Atta S.K., Mohammed S.A., van Cleemput O., Zayed A. (1996) Transformations of iron and manganese under controlled Eh, Eh-pH conditions and additions of organic matter, Soil Technol. 9, 223–237.
- Bagci S.A., Ekiz H., Yilmaz A., Cakmak I. (2007) Effects of zinc deficiency and drought on grain yield of field-grown wheat cultivars in Central Anatolia, J. Agron. Crop Sci. 193, 198–206.
- Baligar V.C., Duncan R.R., Fageria N.K. (1990) Soil-plant interactions on nutrient use efficiency in plants. An overview, in: Baligar V.C., Duncan R.R. (Eds.), Crops as enhancers of nutrient use, Academic Press, San Diego, California, pp. 351–373.
- Baligar V.C., Fageria N.K., He Z.L. (2001) Nutrient use efficiency in plants, Commun. Soil Sci. Plant Anal. 32, 921–950.
- Barker A.V., Pilbeam D.J. (2007) Handbook of Plant Nutrition, Taylor and Francis Group Press, Boca Raton, FL.
- Barnes J.P., Putnam A.R. (1986) Evidence for allelopathy by residues and aqueous extracts of rye (*Secale cereale* L.), Weed Sci. 34, 384–390.
- Barrier-Guillot B., Casado P., Maupetit P., Jondreville C., Gatel F. (1996) Wheat phosphorus availability. In vitro Study; Factors affecting endogenous phytasic activity and phytic phosphorus content, J. Sci. Food Agr. 70, 62–68.
- Bhowmik P.C., Doll J.D. (1984) Allelopathic effects of annual weed residues on growth and nutrient uptake of corn and soybeans, Agron. J. 76, 383–388.
- Black R. (2003) Micronutrient deficiency an underlying cause for morbidity and mortality, B. World Health Organ. 2003, 81–79.
- Blair G. (1993) Nutrient efficiency-what do we really mean, in: Randall P.J., Delhaize E., Richards R.A., Munns R. (Eds.), Genetic aspects of mineral nutrition, Kluwer Academic Publishers, Dordrecht, The Netherlands, pp. 205–213.
- Blum A. (1988) Plant Breeding for Stress environments, CRC Press, Florida, p. 212.
- Borlaug N.E., Dowswell C.R. (1993) Fertilizer: To nourish infertile soil that feeds a fertile population that crowds a fragile world, Fertil. News 387, 11–20.
- Bouis H. (1996) Enrichment of food staples through plant breeding: A new strategy for fighting micronutrient malnutrition, Nutr. Rev. 54, 131–137.
- Brabin B.J., Coulter J.B.S. (2003) Nutrition-associated disease, in: Cook G.C., Zumla A.I. (Eds.), Manson's tropical diseases, Saunders, London, pp. 561–580.
- Brady N.C., Weil R.R. (2002) The nature and properties of soils, 13th edition, Prentice Hall, Upper Saddle River, New Jersey.
- Brennan R.F. (1992) Effect of zinc fertilizer on take-all and grain yield of wheat grown on zinc deficient soils of the esperance region, West. Aust. Fer. Res. 31, 215–216.
- Brennan R.F., Bolland M.D.A. (2006) Residual values of soilapplied zinc fertiliser for early vegetative growth of six crop species, Aust. J. Exp. Agr. 46, 1341–1347.
- Broadley M.R., White P.J., Bryson R.J., Meacham M.C., Bowen H.C., Johnson S.E., Hawkesford M.J., McGrath S.P., Zhao

F.J., Breward N., Harriman M., Tucker M. (2006) Biofortification of UK food crops with selenium, Proc. Nutr. Soc. 65, 169–181.

- Bronick C.J., Lal R. (2005) Manuring and rotation effects on soil organic carbon concentration for different aggregate size fractions on two soils in northeastern Ohio, USA, Soil Till. Res. 81, 239–252.
- Brown L.R. (1997) Facing the challenge of food scarcity: Can we raise grain yields fast enough? in: Ando T., Fujita K., Mae T., Matsumoto H., Mori S., Sekiya J. (Eds.), Plant nutrition for sustainable food production and environment, Kluwer Academic Publishers, Dordrecht, The Netherlands, pp. 15–24.
- Buso G.S.C., Bliss F.A. (1988) Variability among lettuce cultivars grown at two levels of available phosphorus, Plant Soil 111, 67–73.
- Buyckx M. (1993) The international community's commitment to combating micronutrient deficiencies, Food Nutr. Agr. 7, 2–7.
- Byrnes B.H., Bumb B.L. (1998) Population growth, food production and nutrient requirements, in: Rengel Z. (Ed.), Journal of Crop Production. The Haworth Press, New York, pp. 1–27.
- Cai S., Yue L., Jin T., Nordberg G. (1998) Renal dysfunction from cadmium contamination of irrigation water: Doseresponse analysis in a Chinese population, B. World Health Organ. 76, 153–159.
- Cakmak I. (2002) Plant nutrition research: Priorities to meet human needs for food in sustainable ways, Plant Soil 247, 3–24.
- Cakmak I. (2008) Enrichment of cereal grains with zinc: Agronomic or genetic biofortification? Plant Soil 302, 1–17.
- Cakmak I., Marschner H. (1986) Mechanism of phosphate induced zinc deficiency in cotton. I. Zinc deficiency-enhanced uptake rate of phosphorus, Physiol. Plant 68, 483–490.
- Cakmak I., Sari N., Marschner H., Ekiz H., Kalayci M., Yilmaz A., Braun H.J. (1996) Phytosiderophore release in bread and durum wheat genotypes differing in zinc efficiency, Plant Soil 180, 183–189.
- Cakmak I., Ekiz H., Yilmaz A., Torun B., Koleli N., Gultekin I., Alkan A., Eker S. (1997a) Differential response of rye, triticale, bread and durum wheats to zinc deficiency in calcareous soils, Plant Soil 188, 1–10.
- Cakmak I., Derici R., Torun B., Tolay I., Braun H.J., Schlegel R. (1997b) Role of rye chromosomes in improvement of zinc efficiency in wheat and triticale, Plant Soil 196, 249–253.
- Cakmak I., Torun B., Erenoglu B., Oztürk L., Marschner H., Kalayci M., Ekiz H., Yilmaz A. (1998) Morphological and physiological differences in cereals in response to zinc deficiency, Euphytica 100, 349–357.
- Caradus J.R. (1990) Mechanisms improving nutrient use by crop and herbage legumes, in: Baligar V.C., Duncan R.R. (Eds.), Crops as enhancers of nutrient use, San Diego, Academic Press, California, pp. 253–311.
- Carr A.J.H., Stoddart J.L. (1963) The ameliorating effect of zinc on symptoms of phyllody virus in white clover, Ann. Appl. Biol. 51, 259–268.
- Cary E.E., Norvell W.A., Grunes D.L., Welch R.M., Reid W.S. (1994) Iron and Manganese Accumulation by the brz Pea Mutant Grown in Soils, Agron. J. 86, 938–41.
- Chaney R.L., Ryan J.A. (1993) Heavy metals and toxic organic pollutants in MSW-composts: Research results on phy-

toavailability, bioavailability, etc., in: Hoitink H.A.J., Keener H.M. (Eds.), Science and Engineering of Composting: Design, Environmental, Microbiological and Utilization Aspects, Ohio State Univ., Columbus, OH, pp. 451–506.

- Chaney R.L., Reeves P.G., Ryan J.A., Simmons R.W., Welch R.M., Angle J.S. (2004) An improved understanding of soil Cd risk to humans and low cost methods to remediate soil Cd risks, BioMetals 17, 549–553.
- Christian P., West K.P. Jr. (1998) Interactions between zinc vitamin A: An update, Am. J. Clin. Nutr. 68 (Suppl.), S435–S441.
- Clark R.B. (1990) Physiology of cereals for mineral nutrient uptake, use, and efficiency, in: Baligar V.C., Duncan R.R. (Eds.), Crops as enhancers of nutrient use, Academic Press, San Diego, California, pp. 131–209.
- Cobiac L., Baghurst K. (1993) Iron status and dietary iron intakes of Australians, CSIRO Astralia, Adelaide, SA.
- Combs G.F. Jr., Welch R.M. (1998) Creating Healful Food Systems: Linking Agriculture to Human Needs, Cornell International Institute for Food, Agriculture and Development, Cornell University, Ithaca, NY, pp. 1–34.
- Combs G.F. Jr., Welch R.M., Duxbury J.M., Uphoff N.T., Nesheim M.C. (1996) Food-Based Approaches to Preventing Micronutrient Malnutrition: an International Research Agenda. Cornell International Institute for Food, Agriculture, and Development, Cornell University, Ithaca, NY, pp. 1–68.
- Cottenie A., Kang B.T., Kieken, S.L., Sajjapongse A. (1981) Micronutrient status, in: Greenland D.J. (Ed.), Characterization of Soils, Clarendon, Oxford, UK, pp. 149–163.
- DellaPenna D. (1999) Nutritional genomics: manipulating plant micronutrients to improve human health, Science 285, 375– 379.
- Diekow J.J., Mielniczuk J., Knicker H., Bayer C., Dick D.P., Kögel-Knabner I. (2005) Soil C and N stocks as affected by cropping systems and nitrogen fertilisation in a southern Brazil Acrisol managed under no-tillage for 17 years, Soil Till. Res. 81, 87–95.
- Donald C.M., Prescott J.A. (1975) Trace elements in Australian crop and pasture production, in: Nicholas D.J.D., Egan R.A. (Eds.), Trace Elements in Soil-Plant-Animal Systems, Academic Press, New York, pp. 7–37.
- Drakakaki G., Marcel S., Glahn R.P., Lund E.K., Pariagh S., Fischer F., Christou P., Stoger E. (2005) Endosperm-specific coexpression of recombinant soybean ferritin and Aspergillus phytase in maize results in significant increases in the levels of bioavailable iron, Plant Mol. Biol. 59, 869–880.
- Eghball B., Ginting D., Gilley J.E. (2004) Residual effects of manure and compost applications on corn production and soil properties, Agron. J. 96, 442–447.
- Erdal I., Yilmaz A., Taban S., Eker S., Torun B., Cakmak I. (2002) Phytic acid and phosphorus concentrations in seeds of wheat cultivars grown with and without zinc fertilization, J. Plant Nutr. 25, 113–127.
- Fageria N.K. (2000a) Adequate and toxic levels of boron for rice, common bean, corn, soybean and wheat production in Cerrado soil, R. Bras. Eng. Agr. Ambiental. 4, 57–62.
- Fageria N.K. (2000b) Adequate and toxic levels of zinc for rice, common bean, corn, soybean and wheat production in Cerrado soil, R. Bras. Eng. Agr. Ambiental. 4, 390–395.

- Fageria N.K., Baligar V.C. (2003) Methodology for evaluation of lowland rice genotypes for nitrogen use efficiency, J. Plant Nutr. 26, 1315–1333.
- Fageria N.K., Barbosa Filho M.P. (2001) Nitrogen use efficiency in lowland rice genotypes, Commun. Soil Sci. Plan. 32, 2079–2089.
- Fageria N.K., Baligar V.C., Clark R.B. (2002) Micronutrients in crop production, Adv. Agron. 77, 185–268.
- Fageria N.K., Baligar V.C., Li Y.C. (2008) The role of nutrient efficient plants in improving crop yields in the twenty first Century, J. Plant Nutr. 31, 1121–1157.
- FAO (Food and Agriculture Organization) (1999) Yearbook of Fertilizers, Food and Agriculture Organization of the United Nations, Rome, Italy.
- Fernandez G.C.J. (1991) Analysis of cultivar × environment interaction by stability estimates, HortScience 26, 947–950.
- Fernandez G.C.J., Chen H.K., Miller J.C. Jr. (1989) Adaptation and environmental sensitivity of mungben genotypes evaluated in the International Mungbean Nursery, Euphytica 41, 253–261.
- Fischer R.A., Maurer R. (1978) Drought resistance in spring wheat cultivars. I. Grain yield responses, Aust. J. Agr. Res. 29, 897–912.
- Fleming G.A., Mordenti A. (1991) The Production of Animal Wastes, European Conference on Environment and Agriculture, Stock Farming in Europe, Mantua, Italy.
- Food and Agriculture Organization of the United Nations (2004) Undernourishment around the world, in: The state of food insecurity in the world 2004, The Organization, Rome.
- Foy C.D. (1984) Physiological effects of hydrogen, aluminum and manganese toxicities in acid soils, in: Adams F. (Ed.), Soil Acidity and Liming, 2nd ed., American Society of Agronomy, Madison, WI, pp. 57–97.
- Foy C.D. (1992) Soil chemical factors limiting plant growth, Limitations to Plant Root Growth, in: Hatfield J.L., Stewart B.A. (Eds.), Adv. Soil Sci. 19, 97–149.
- Frey K.J. (1964) Adaption reaction of oat strains selected under stress and non-stress environmental condition, Crop Sci. 4, 55–58.
- Frossard E., Bucher M., Mächler F., Mozafar A., Hurrell R. (2000) Potential for increasing the content and bioavailability of Fe, Zn and Ca in plants for human nutrition, J. Sci. Food Agr. 80, 861–879.
- Galrão E.Z. (1999) Methods of copper application and evaluation of its availability for soybean grown on a Cerrado redyellow latosol, R. Bras. Ci. Solo. 23, 265–272.
- Gao M., Che F.C., Wei C.F., Xie D.T., Yang J.H. (2000) Effect of long-term application of manures on forms of Fe, Mn, Cu and Zn in purple paddy soil, Plant Nutr. Fertil. Sci. 6, 11–17.
- Garcia-Casal M.N., Layrisse M., Solano L., Baron M.A., Arguello F., Llovera D., Ramirez J., Leets I., Tropper E. (1998) Vitamin A and beta carotene can improve nonheme iron absorption from rice, wheat and corn by humans, J. Nutr. 128, 646–650.
- Gargari B.P., Mahboob S., Razavieh S.V. (2007) Content of phytic acid and its mole ratio to zinc in flour and breads consumed in Tabriz, Iran, Food Chem. 100, 1115–1119.
- Gerloff, Gabelman (1983) Genetic basis of inorganic plant nutrition, in: Lauchli A., Bieleski R.L. (Eds.), Inorganic Plant Nutrition. Encyclopedia and Plant Physiology New Series, Springer Verlag, New York, NY, Vol. 15B, pp. 453–480.

- Gibson R.S. (1994) Zinc nutrition in developing countries, Nutr. Res. Rev. 7, 151–173.
- Gibson R.S. (2006) Zinc: the missing link in combating micronutrient malnutrition in developing countries, Proc. Nutr. Soc. 65, 51–60.
- Gibson R.S., Sazawal S., Peerson J.M. (2003) Design and quality control issues related to dietary assessment, randomized clinical trials and meta-analysis of field-based studies in developing countries, J. Nutr. 133 (5 Suppl. 1), 1569s–1573s.
- Gibson R.S., Hess S.Y., Hotz C., Brown. K.H. (2008) Indicators of zinc status at the population level: a review of the evidence, Brit. J. Nutr. 99 (Suppl. 3), S14–S23.
- Glass A.D.M. (1989) Physiological mechanisms involved with genotypic differences in ion adsorption and utilization, HortScience 24, 559–564.
- Godwin D.C., Blair G.J. (1991) Phosphorus efficiency in pasture species. V. A comparison of white clover accessions, Aust. J. Agr. Res. 42, 531–540.
- Gonzalez D., Obrador A., Alvarez J.M. (2007) Behavior of zinc from six organic fertilizers applied to a navy bean crop grown in a calcareous soil, J. Agr. Food Chem. 55, 7084–7092.
- Goto F., Yoshihara T., Shigemoto N., Toki S., Takaiwa F. (1999) Iron fortification of rice seed by the soybean ferritin gene, Nat. Biotechnol. 17, 282–286.
- Goto K., Tanimoto Y., Tamura T., Mochida K., Arai D., Asahara M., Suzuki M., Tanaka H., Inagaki K. (2002) Identification of thermoacidophilic bacteria and a new *Alicyclobacillus* genomic species isolated from acidic environments in Japan, Extremophiles 6, 333–340.
- Gourley C.J.P., Allan D.L., Russelle M.P. (1994) Plant nutrient efficiency: A comparison of definitions and suggested improvement, Plant Soil 158, 29–37.
- Graham R.D. (1984) Breeding for nutritional characteristics in cereals, in: Tinker P.B., Lauchi A. (Eds.), Advances in plant nutrition, Vol. 1, Praeger Publisher, New York, pp. 57–102.
- Graham R.D., Webb M.J. (1991) Micronutirents and disease resistance and tolerance in plants, in: Mortvadt J.J., Cox F.R., Shuman L.M., Welch R.M. (Eds.), Micronutrients in Agriculture, 2nd ed., Soil Scienc Society American Book series No. 4, Madison, WI, pp. 329–370.
- Graham R.D., Rengel Z. (1993) Genotypic variation in zinc uptake and utilization by plants, in: Robson A.D. (Ed.), Zinc in Soils and Plants, Kluwer Academic Publishers, Dordrecht, pp. 107–118.
- Graham R.D., Welch R.M. (1996) Breeding for staple-food crops with high micronutrient density. Agricultural Strategies for Micronutrients. Working Paper No. 3, International Food Policy Research Institute, Washington, DC.
- Graham R.D., Ascher J.S., Hynes S.C. (1992) Selecting zinc efficient cereal genotypes for soils of low zinc status, Plant Soil 146, 241–250.
- Graham R.D., Senadhira D., Beebe S.E., Iglesias C., Ortiz-Monasterio I. (1999) Breeding for micronutrients density in edible portions of staple food crops: conventional approaches, Field Crop. Res. 60, 57–80.
- Graham R.D., Humphries J.M., Kitchen J.L. (2000) Nutritionally enhanced cereals: A sustainable foundation for a balanced diet, Asia Pac. J. Clin. Nutr. 9 (Suppl.), S91–S96.
- Graham R.D., Welch R.M., Bouis H.E. (2001) Addressing micronutrient malnutrition through enhancing the nutritional

quality of staple foods: Principles, perspectives and knowledge gaps, Adv. Agron. 70, 77–142.

- Grant C.A., Clarke J.M., Duguid S., Chaney R.L. (2008) Selection and breeding of plant cultivars to minimize cadmium accumulation, Sci. Total Environ. 390, 301–310.
- Grewal H.S., Graham R.D. (1999) Residual effects of subsoil zinc and oilseed rape genotype on the grain yield and distribution of zinc in wheat, Plant Soil 207, 29–36.
- Gunes A., Inal A., Adak M.S., Alpaslan M., Bagci E.G., Erol T., Pilbeam D.G. (2007) Mineral nutrition of wheat, chickpea and lentil as affected by mixed cropping and soil moisture, Nutr. Cycl. Agroecosys. 78, 83–96.
- Hacisalihoglu G. (2002) Physiological and biochemical mechanisms underlying zinc efficiency in monocot and dicot crop plants, PhD Thesis, Cornell University, Ithaca, New York, USA.
- Harris D., Rashid A., Miraj Gh., Arif M., Yunas M. (2008) 'On-farm' seed priming with zinc in chickpea and wheat in Pakistan, Plant Soil 306, 3–10.
- Harris D., Rashid A., Miraj Gh., Arif M., Shah H. (2007) 'Onfarm' seed priming with zinc in chickpea and wheat in Pakistan, Plant Soil 306, 3–10.
- Hart J.J., Welch R.M., Norvell W.A., Kochian L.V. (2002) Transport interactions between cadmium and zinc in roots of bread and durum wheat seedlings, Physiol. Plantarum 116, 73–78.
- Hill C.H., Matrone G. (1970) Chemical parameters in the study of in vivo and in vitro interactions of transition elements, Fed Proc. 29, 1474–1481.
- Hodges R.E., Saurberlich H.E., Canham J.E., Wallace D.L., Rucker R.B., Mejia L.A., Mohanram M. (1978) Hematopoietic studies in vitamin A deficiency, Am. J. Clin. Nutr. 31, 876–885.
- House W.A. (1999) Trace element bioavailability as exemplified by iron and zinc, Field Crop. Res. 60, 115–141.
- Huber D.M., Abney, T.S. (1986) Soybean allelopathy and subsequent cropping, J. Agron. Crop Sci. 157, 73–78.
- Hylander L.D. (1995) Effects of lime, phosphorus, manganese, copper and zinc on plant mineral composition, yield of barley, and level of extractable nutrients for an acid Swedish mineral soil, Commun. Soil Sci. Plan. 26, 2913–2928.
- International Zinc Nutrition Consultative Group (IZINCG) (2004) Assessment of the risk of zinc deficiency in populations and options for its control, in: Hotz C., Brown K.H. (Eds.), Food and nutrition bulletin, Vol. 25, No. 1, Suppl. 2, pp. S113–S118.
- Isfan D. (1993) Genotypic variability for physiological efficiency index of nitrogen in oats, Plant Soil 154, 53–59.
- Kalayci M., Torun B., Eker S., Aydin M., Ozturk L., Cakmak I. (1999) Grain yield, zinc efficiency and zinc concentration of wheat cultivars grown in a zinc-deficient calcareous soil in field and greenhouse, Field Crop. Res. 63, 87–98.
- Kang Y.L. (1988) Effects of organic matter and its anaerobic decomposition products on the growth and zinc uptake by *Oryza sativa*, Disser. Abst. Int. B (Sci., Eng.) 49, 953B.
- Kannenberg L.W., Falk D.E. (1995) Models for activation of plant genetic resources for crop breeding programs, Can. J. Plant Sci. 75, 45–53.
- Katyal J.C. (1977) Influence of organic matter on the chemical and electrochemical properties of some flooded soils, Soil Biol. Biochem. 9, 259–266.

- Katyal J.C., Vlek P.L.G. (1985) Micronutrient problems in tropical Asia, Fert. Res. 7, 69–94.
- Keller A., Schulin R. (2003) Modelling heavy metal and phosphorus balances for farming systems, Nutr. Cycl. Agroecosys. 66, 271–284.
- Khoshgoftarmanesh A.H., M. Kalbasi (2002) Effect of Municipal waste Leachate on soil properties and growth and yield of rice, Commun. Soil Sci. Plan. 33, 2011–2020.
- Khoshgoftar, A.H., HajiMozaffari E. (2006) Approaches to enhance iron concentration in wheat grain produced in Qom province. Iran. 2nd Central Asian Cereals Conference, June 13–16, 2006, Cholpon-Ata, Issyk Kul Lake, Kyrgyz Republic.
- Khoshgoftar A.H., Chaney R.L. (2007) Preceding crop affects grain cadmium and zinc of wheat grown in saline soils of central Iran, J. Environ. Qual. 36, 1132–1136.
- Khoshgoftarmanesh A.H., Shariatmadari H., Kalbasi M., Karimian N. (2004a) Zinc efficiency of wheat cultivars grown on a saline calcareous soil, J. Plant Nutr. 27, 1953–1962.
- Khoshgoftar A.H., Shariatmadari H., Karimian N., Kalbasi M., van der Zee S.E.A.T.M., Parker D.R. (2004b) Salinity and Zn application effects on phytoavailability of Cd and Zn, Soil Sci. Soc. Am. J. 68, 1885–1889.
- Khoshgoftar A.H., Shariatmadari H., Karimian N., and van der Zee S.E.A.T.M. (2006a) Cadmium and zinc in saline soil solutions and their concentrations in wheat, Soil Sci. Soc. Am. J. 70, 582–589.
- Khoshgoftar A.H., Shariatmadari H., Karimian N. (2006b) Responses of wheat genotypes to zinc fertilization under saline soil conditions, J. Plant Nutr. 27, 1–14.
- Khoshgoftarmanesh A.H., Sharifi H.R., Mirzapour M.H., Schulin R. (2007) Plant genotype and Zn fertilization effects on nutritional quality of wheat grain produced in saline soils. 9th International Conference of the Biochemistry of Trace Elements (ICOBTE), July 2007, Beijing, China.
- Kochian L.V. (1995) Cellular mechanisms of aluminum toxicity and resistance in plants, Annu. Rev. Plant Phys. 46, 237–260.
- Leon L.A., Lopez A.S., Vlek P.L.G. (1985) Micronutrient problems in tropical Latin America, Fert. Res. 7, 95–129.
- Liu X.J., Liao X.Y., Zhang Y.Z., Huang Y.X. (2002) Effects of ricebased cropping system on the distribution of manganese in the profile of paddy soil derived from red earth, Acta Ecol. Sin. 22, 1440–1445.
- Loneragen J.P. (1997) Plant nutrition in 20th and perspectives for the 21st century, Plant Soil 196, 163–174.
- Lopes A.S., Cox F.R. (1977) A survey of the fertility status of surface soils under 'Cerrado' vegetation in Brazil, Soil Sci. Soc. Am. J. 41, 742–747.
- Lucca P., Hurrell R.F., Potrykus I. (2001) Genetic engineering approaches to improve the bioavailability and the level of iron in rice grains, Theor. Appl. Genet. 102, 392–397.
- Marschner H. (1995) Mineral Nutrition of Higher Plants, 2nd ed., Academic Press, London.
- Mason J.B., Garcia M. (1993) Micronutrient deficiency the global situation, SCN News 9, 11–16.
- Martens D.C., Westerman D.T. (1991) Fertilizer applications for correcting micronutrient deficiencies, in: Mortvedt J.J., Cox F.R., Shuman L.M., Welch R.M. (Eds.), Micronutrients in Agriculture, 2nd ed., SSSA Book Series 4, SSSA, Madison, WI, pp. 549–592.

- Martens D.C., Lindsay W.L. (1990) Testing soils for copper, iron, manganese, and zinc, in: Westerman R.L. (Ed.), Soil Testing and Plant Analysis, Soil Science Society of America, Madison, WI, pp. 229–273.
- Mayer J.E., Pfeiffer W.H., Beyer P. (2008) Biofortified crops to alleviate micronutrient malnutrition, Curr. Opin. Plant Biol. 11, 166–170.
- McGuire J. (1993) Addressing micronutrient malnutrition, SCN News 9, 1–10.
- McIntyre B.D., Bouldin D.R., Urey G.H., Kizito F. (2001) Modeling cropping strategies to improve human nutrition in Uganda, Agr. Syst. 67, 105–120.
- McLaughlin M.J., Parker D.R., Clarke J.M. (1999) Metals and micronutrients - Food safety issues, Field Crop. Res. 60, 143–163.
- Meharg A.A. (2004) Arsenic in rice understanding a new disaster for South-East Asia, Trends Plant Sci. 9, 415–417.
- Menzi H., Kessler J. (1998) Heavy metal content of manures in Switzerland, in: Proceedings of the Eighth International Conference of the FAO Network on Recycling of Agricultural, Municipal and Industrial Residues in Agriculture (in press).
- Miller D., Waissman N., Melton B., Currier C., McCaslin B. (1987) Selection for increased phosphorus in alfalfa and effects on other characteristics, Crop Sci. 27, 22–26.
- Mitchell C.C., Entry J.A. (1998) Soil C, N and crop yields in Alabama's long-term 'old rotation' cotton experiment, Soil Till. Res. 47, 331–338.
- Mortvedt J.J. (1985) Plant uptake of heavy metals in zinc fertilizers made from industrial by-products, J. Environ. Qual. 14, 424–427.
- Mortvedt J.J. (1994) Needs for controlled-availability micronutrient fertilizers, Fert. Res. 38, 213–221.
- Mortvedt, J.J. (1996) Heavy metal contaminants in inorganic and organic fertilizers, Fert. Res. 43, 55–61.
- Murray C.J.L., Lopez A.D. (1997) Global mortality, disability, and the contribution of risk factors: Global Burden of Disease Study, Lancet 349, 1436–42.
- Nable R.O., Webb M.J. (1993) Further evidence that zinc is required throughout the root zone for optimal growth and development, Plant Soil 150, 247–253.
- Neidecker-Gonzales O., Nestel P., Bouis H. (2007) Estimating the global costs of vitamin A capsule supplementation: A review of the literature, Food Nutr. Bull. 28, 307–316.
- Nicholson F.A., Chambers B.J., Williams J.R., Unwin R.J. (1999) Heavy metal contents of livestock feeds and animal manures in England and Wales, Bioresource Technol. 70, 23– 31.
- Nicklas T.A. (1995) Dietary studies of children: the Bogalu Heart Study experience, J. Am. Diet. Assoc. 95, 1127–1133.
- Nogawa K., Kobayashi E., Okubo Y., Suwazono Y. (2004) Environmental cadmium exposure, adverse effects and preventive measures in Japan, BioMetals 17, 581–587.
- Novack B., Schwyzer I., Schulin R. (2008) Uptake of Zn and Fe by Wheat (*Triticum aestivum* var. Greina) and Transfer to the Grains in the Presence of Chelating Agents (Ethylenediaminedisuccinic Acid and Ethylenediaminetetraacetic Acid), Agr. Food Chem. 56, 4643–4649.
- O'Dell B.L., de Boland A.R., Koirtyohann S.R. (1972) Distribution of phytate and nutritionally important elements among the morphological components of cereal grains, J. Agr. Food Chem. 20, 718–721.

- Pareek S., Pareek S. (1999) Effect of macro-and micro-nutrients on charcoal rot disease development of maize induced by Macrophominea phaseolina, Ann. Agr. Res. 20, 129–131.
- Pessarakli M. (ed.) (1999) Handbook of plant and crops stress, Marcel Dekker, New York.
- Pinstrup-Andersen P. (1999) Selected aspects of the future global food situation, in: 25th International Fertilizer Industry Association Enlarged Council Meeting, Rome, Italy.
- Prasad B. (1999) Conjoint use of fertilizers with organics, crop residues and green manuring for their efficient use in sustainable crop production, Fert. News 44, 67–73.
- Prasad B., Sinha M.K. (1981) The relative efficiency of zinc carriers on growth and zinc nutrition of corn, Plant Soil 62, 45–52.
- Prasad B., Sinha S.K. (1995a) Effect of recycling of crop residues and organic manure on capacity factor and diffusion rate of zinc in calcareous soil, J. Nucl. Agr. Biol. 24, 185–188.
- Prasad B., Sinha S.K. (1995b) Nutrient recycling through crop residues management for sustainable rice and wheat production in calcareous soil, Fert. News 40, 15–2325.
- Raboy V., Noaman M.W., Taylor G.A., Pickett S.G. (1991) Grain phytic acid and protein are highly correlated in winter wheat, Crop Sci. 31, 631–635.
- Raj H., Gupta V.K. (1986) Influence of organic manures and zinc on wheat yield and Zn concentration in wheat, Agr. Wastes 16, 255–263.
- Ramalingaswami V. (1995) New global perspectives on overcoming malnutrition, Am. J. Clin. Nutr. 61, 259–263.
- Reeves P.G. Chaney R.L. (2008) Bioavailability as an issue in risk assessment and management of food cadmium: A review, Sci. Total Environ. 398, 13–19.
- Rengel Z. (2001) Genotypic differences in micronutrient use efficiency in crops, Commun. Soil Sci. Plan. 32, 1163–1186.
- Rengel Z., Graham R.D. (1995) Importance of seed Zn content for wheat growth on Zn-deficient soil. 2. Grain yield, Plant Soil 173, 267–274.
- Rengel Z., Batten G.G., Crowley D.E. (1999) Agronomic approaches for improving the micronutrient density in edible portions of field crops, Field Crop. Res. 60, 27–40.
- Reuter D.J., Alston A.M., McFarland J.D. (1988) Occurrence and correction of manganese deficiency in plants, in: Graham R.D., Hannan R.J., Uren N.C. (Eds.), Manganese in Soils and Plants, Kluwer Academic, Dordrecht, The Netherlands, pp. 205–224.
- Rice E.L. (1984) Allelopathy, 2nd ed, Academic Press, Orlando, FL, 422 p.
- Rose S.J., Burnside O.C., Specht J.E., Swisher B.A. (1984) Competition and allelopathy between soybeans and weeds, Agron. J. 76, 523–528.
- Rosegrant M.W., Leach N., Gerpacio R.V. (1999) Alternative future for world cereal and meat consumption, Proc. Nutr. Soc. 58, 219–234.
- Rosegrant M.W., Paisner M.S., Meijer S., Witcover J. (2001) 2020 Global Food Outlook; Trends, Alternatives, and Choices. A 2020 Vision for Food, Agriculture, and the Environment Initiative, International Food Policy Research Institute, Washington, DC.
- Rosielle A.A., Hamblin J. (1981) Theoretical aspects of selection for yield in stress and non-stress environments, Crop Sci. 21, 943–946.

- Sanghvi T.G. (1996) Economic Rationale for Investing in Micronutrient Programs. A Policy Brief Based on New Analyses. Office of Nutrition, Bureau for Research and Development, United States Agency for International Development, Washington, DC.
- Sauerbeck D.R., Helal H.M. (1990) Factors affecting the nutrient efficiency in plants, in: El Balsam N., Dambroth M., Loughman B.C. (Eds.), Genetic Aspects of Plant Mineral Nutrition, Kluwer Academic Publisher, Dordrecht the Netherlands, pp. 11–17.
- Schulin R., Khoshgoftarmanesh A., Afyuni M., Nowack B., Frossard E. (2009) Effect of soil management on zinc uptake and its bioavailability in plants, in: Banuelos G.S., Lin Z. (Eds.), Development and uses of biofortified agricultural products, CRC Press, Boca Raton, FL.
- Shaver T.M., Westfall D.G., Ronaghi M. (2007) Zinc fertilizer solubility and its effects on zinc bioavailability over time, J. Plant Nutr. 30, 123–133.
- Siddiqi M.Y., Glass A.D. (1981) Utilization Index: A modified approach to the estimation and comparison of nutrient utilization efficiency in plants, J. Plant Nutr. 4, 289–302.
- Sillanpaa M. (1982) Micronutrients and Nutrient Status of Soils: A Global Study; Food and Agriculture Organization: Rome, FAO Soil Bull. No. 48.
- Sillanpaa M., Vlek P.L.G. (1985) Micronutrients and the agroecology of tropical and Mediterranean regions, Fert. Res. 7, 151–168.
- Sims J.T., Wolf D.C. (1994) Poultry waste management: Agricultural and Environmental issues, Adv. Agron. 52, 2–72.
- Singh G., Abrol I.P., Cheema S.S. (1989) Effects of gypsum application on mesquite (Prosopisjulitlora) and soil properties in an abandoned sodic soil, Forest. Ecol. Manag. 29, 1–14.
- Singh B., Natesan S.K.A., Singh B.K. Usha K. (2005) Improving zinc efficiency of cereals under zinc deficiency, Curr. Sci. 88, 36–44.
- Soil Science Society of America (1997) Glossary of soil science terms, Soil Science Society of America, Madison, Wisconsin.
- Solomons N.W., Russell R.M. (1980) The interaction of vitamin A and zinc: Implications for human nutrition, Am. J. Clin. Nutr. 33, 2031–2040.
- Stein A.J. (2006) Micronutrient Malnutrition and the Impact of Modern Plant Breeding on Public Health in India: How Costeffective is Biofortification? Cuvillier Verlag.
- Stein A.J., Nestel P., Meenakshi J.V., Qaim M., Sachdev H.P., Bhutta Z.A. (2007) Plant breeding to control zinc deficiency in India: how cost-effective is biofortification? Public Health Nutr. 10, 492–501.
- Stevenson F.J. (1991) Organic matter-micronutrient reaction in soil, in: Mortvedt J.J. et al. (Eds.), Micronutrients in Agriculture, Soil Sci. Soc. Am., Madison, WI, pp. 145–186.
- Stevenson F.J. (1994) Humus chemistry; Genesis composition. Reaction, Wiley New York, p. 496.
- Stewart W.M., Dibb D.W., Johnston A.E., Smyth T.J. (2005) The contribution of commercial fertilizer nutrients to food production, Agron. J. 97, 1–6.
- Streeter J.G. (2001) Simple partial purification of D-pinitol from soybean leaves, Crop Sci. 41, 1985–1987.
- Swaddiwudhipong W., Limpatanachote P., Mahasakpan P., Krintratun S., Padungtod C. (2007) Cadmium-exposed population in Mae Sot District, Tak Province: 1. Prevalence of

high urinary cadmium levels in the adults, J. Med. Assoc. Thai 90, 143–148.

- Tagwira F. (1993) Some chemical and physical factors that associated with micronutrient status and availability in Zimbabwean soils, Zimbabwean J. Agr. Res. 31, 1051–1059.
- Takkar P.N., Walker C.D. (1993) The distribution and correction of zinc deficiency, in: Robson A.D. (Ed.), Zinc in Soils and Plants, Kluwer Academic Publishers, Dordrecht, The Netherlands, pp. 151–165.
- Thung M. (1988) Phosphorus: A limiting nutrient in bean (*Phaseolus vulgaris* L.) production in Latin America and field screening for efficiency and response, in: Bassam N., Dambroth M.C., Loughman B.C. (Eds.), Genetic Aspects of Plant Mineral Nutrition, Kluwer Academic Publishers, Dordrecht, pp. 501–521.
- Thongbai P., Hannam R.J., Graham R.D., Webb M.J. (1993) Interaction between zinc nutritional status of cereals and Rhizoctonia root rot severity. I. Field observation, Plant Soil 153, 207–214.
- Tillman D. (1999) Global environmental impacts of agricultural expansion: the need for sustainable and efficient practices, Proc. Natl Acad. Sci. (USA) 96, 5995–6000.
- Timsina J., Connor D.J. (2001) Productivity and management of rice-wheat cropping systems: Isssues and challenges, Field Crop. Res. 69, 93–132.
- Torun A., Gültekin I., Kalayci M., Yilmaz A., Eker S., Cakmak I. (2001) Effects of zinc fertilization on grain yield and shoot concentrations of zinc, boron, and phosphorus of 25 wheat cultivars grown on a zinc-deficient and boron-toxic soil, J. Plant Nutr. 24, 1817–1829.
- UNICEF: Vitamin A Supplementation: A Decade of Progress (2007) The United Nations Children's Fund (UNICEF).
- United Nations General Assembly (2000) United Nations Millennium Declaration, A/RES/55/2.
- United Nations Administrative Committee on Coordination Subcommittee on Nutrition Situation (1992) Vol. 1. Global and Regional Results, World Health Organization, Geneva, Switzerland, pp. 1–80.
- US Department of Health and Human Services. Healthy people 2000, Nutrition Today (1990) November/December, 29–30.
- Verna T.S., Minhas R.S. (1987) Zinc and phosphorus interaction in a wheat-maize cropping system, Fert. Res. 13, 77–86.
- Vijayaraghavan K. (2002) Control of micronutrient deficiencies in India: obstacles and strategies, Nutr. Rev. 60, S73–S76.
- Walker D.W., Jenkins D.D. (1986) Influence of sweet potato plant residue on growth of sweet potato vine cuttings and cowpea plants, HortSci. 21, 426–428.
- Walker D.W., Hubbell T.J., Sedberry J.E. (1989) Influence of decaying sweet potato crop residues on nutrient uptake of sweet potato plants, Agr. Ecosyst. Environ. 26, 45–52.
- Wallace A., Mueller R.T. (1973) Effects of chelated and nonchelated cobalt and copper on yields and microelement composition of bush beans grown on calcareous soil in a glasshouse. Soil Sci. Soc. Am. Proc. 37, 904–908.
- Wallace G.A., Wallace A. (1982) Micronutrient uptake by leaves from foliar sprays of EDTA chelated metals, in: Nelson S.D. (Ed.), Iron Nutrition and Interactions in Plants, Marcel Dekker, Basel, pp. 975–978.
- Webber M.D., Webber L.R. (1983) Micronutrients and heavy metals in livestock and poultry manures, in: Farm Animal Manures in the Canadian Environment. Publication No.

NRCC 18976 of the Environmental Secretariat, Ottawa, Canada, p. 59.

- Wei X., Hao M., Shao M., Gale W.J. (2006) Changes in soil properties and the availability of soil micronutrients after 18 years of cropping and fertilization, Soil Till. Res. 91, 120– 130.
- Welch R.M. (1998) Fashioning healthful agricultural systems, in: Combs G.F. Jr., Welch R.M. (Eds.), Creating Healthful Food Systems: Linking Agriculture to Human Needs Cornell International Institute for Food, Agriculture and Development, Cornell University, Ithaca, NY, pp. 7–13.
- Welch RM. (1999) Importance of seed mineral nutrient reserves in crop growth and development, in: Rengel Z. (Ed.), Mineral Nutrition of Crops: Fundamental Mechanisms and Implications, Food Products Press, New York, pp. 205–226.
- Welch R.M. (2002) The impact of mineral nutrients in food crops on global human health, Plant Soil 247, 83–90.
- Welch R.M., Graham R.D. (1999) A new paradigm for world agriculture: meeting human needs – Productive, sustainable, nutritious, Field Crop. Res. 60, 1–10.
- Welch R.M., Graham R.D. (2000) A new paradigm for world agriculture: productive, sustainable, nutritious, healthful food systems, Food Nutr. Bull. 21, 361–366.
- Welch R.M., Graham R.D. (2004) Breeding for micronutrients in staple food crops from a human nutrition perspective, J. Exp. Bot. 55, 353–364.
- Welch R.M., Allaway W.H., House W.A., Kubota J. (1991) Geographic distribution of trace element problems, in: Mortvedt J.J., Cox F.R., Shuman L.M., Welch R.M. (Eds.), Micronutrients in Agriculture 2nd ed., Soil Sci. Soc. Am., Madison, WI, pp. 31–57.
- Welch R.M., Combs G.F. Jr., Duxbury J.M. (1997) Toward a "Greener" revolution, Issues Sci. Technol. 14, 50–58.
- White P.J., Broadley M.R. (2005) Biofortifying crops with essential mineral elements, Trends Plant Sci. 10, 586–593.
- Williams C.H., David D.J. (1976) Effects of pasture improvement with subterranean clover and superphosphate on the availability of trace metals to plant, Aust. J. Soil Res. 14, 85–93.
- Williams P.N., Price A.H., Raab A., Hossain S.A., Feldmann J., Meharg A.A. (2005) Variation in arsenic speciation and

concentration in paddy rice related to dietary exposure, Environ. Sci. Technol. 39, 5531–5540.

- Woodbury P.B. (1992) Trace elements in municipal solid waste composts: A review of potential detrimental effects on plants, soil biota, and water quality, Biomass Bioenerg. 3, 239–259.
- World Health Organization (1992) Second report on the world nutrition situation. Global and regional results, United Nations Administrative Committee on Coordination – Subcommittee on Nutrition, Geneva, Switzerland, 1, pp. 1–80.
- World Health Organization (1996) Trace elements in human nutrition and health, Geneva, pp. 30–35, 90–91.
- World Health Organization (1999) Malnutrition Worldwide. http://www.who.int/nut/malnutrition\_worldwide.htm, World Health Organization, Geneva, pp. 1–13.
- Yang X.E., Chen W.R., Feng Y. (2007) Improving human micronutrient nutrition through biofortification in the soil-plant system: China as a case study, Environ. Geochem. Hlth 29, 413–428.
- Ye X., Al-Balil S., Kloli A., Zhang J., Lucca P., Beyer P., Potrykus I. (2000) Engineering the provitamine A (b-carotene) biosynthetic pathway into (carotenoid-free) rice endosperm, Science 287, 303–305.
- Yilmaz A., Ekiz H., Torun B. et al. (1997) Effects of different zinc application methods on grain yield and zinc concentrations in wheat grown on zinc-deficient calcareous soils in Central Anatolia, J. Plant Nutr. 20, 461–471.
- Yilmaz A., Ekiz H., Gültekin I., et al. (1998) Effect of seed zinc content on grain yield and zinc concentration of wheat grown in zinc-deficient soils, J. Plant Nutr. 21, 2257–2264.
- Yodkeaw M., De Datta S.K. (1989) Effects of organic matter and water regime on the kinetics of iron and manganese in two high pH rice soils, Soil Sci. Plant Nutr. 35, 323–335.
- Zuo Y., Zhang F. (2008) Effect of peanut mixed cropping with gramineous species on micronutrient concentrations and iron chlorosis of peanut plants grown in a calcareous soil, Plant Soil 306, 23–36.
- Zuo Y., Zhang F. (2009) Iron and zinc biofortification strategies in dicot plants by intercropping with gramineous species. A review, Agron. Sustain. Dev. 29, 63–71.

# Multi-Criteria Decision Models for Management of Tropical Coastal Fisheries

Merlina N. Andalecio

### Contents

1	Intro	luction	252				
2	2 Fundamentals of Multi-Criteria Evaluation						
	Metho	<b>bdb</b> d	254				
	2.1	Choice of Evaluation Criteria and Indicators	255				
	2.2	Measurement of Criteria and Indicators	257				
	2.3	Weights of Importance of Criteria					
		and Indicators	258				
	2.4	Stakeholders and Resource Users					
		as Evaluators	259				
3	Multi	-Criteria Aggregation Approaches	260				
	3.1	Analytic Hierarchy Process (AHP)	260				
	3.2	Weighted Sum Model	262				
	3.3	Ordination Technique	263				
	3.4	Concordance Analysis	264				
	3.5	Regime Method	265				
	3.6	Mixed Data Evaluation (or Evamix)	266				
	3.7	Summary of Multi-Criteria Aggregation					
		Approaches	266				
4	Appli	cation of Multi-Criteria Evaluation Methods					
	to Tro	pical Fisheries Management	268				
	4.1	Case Study: Evaluation of Fisheries					
		Management Strategies in San Miguel Bay,					
		Philippines	268				
	4.2	Extent of Application of Multi-Criteria					
		Methods to Tropical Fisheries Management	273				
5	Concl	usion	274				
Re	ference	es	275				

Abstract The crisis in the world's fisheries is attributed to excessive fishing pressure, long-term mismanagement, increased population growth, development and improvement of fishing technologies, uncertainty in global fisheries catch data, economic incentives and subsidies, and increasing demand for fish meal. For the coastal fisheries of developing countries, the problem is aggravated by coastal habitat degradation, widespread poverty in coastal communities, inshore encroachment of commercial fishing vessels, use of illegal and destructive fishing methods, resource use conflicts, pollution from uplands, and weak institutional arrangements. In response to these problems and in the hope of reversing their negative effects, fisheries management strategies have emerged for tropical coastal fisheries during at least the past 30 years. In fisheries, it is crucial to determine the outcomes of management strategies, especially when public money has to be accounted for. However, efforts to assess their impacts or measure progress are usually directed towards a single disciplinary approach, which fails to consider the multi-dimensionality of tropical fisheries including concomitant multi-level and conflicting goals and objectives. This article explores the utility of a multi-criteria type of evaluation as a potential analytical approach in impact evaluation for tropical fisheries management. The general framework of a multi-criteria evaluation method is a two-dimensional matrix composed of different choice possibilities including the set of criteria and indicators that will serve as bases in assessing these choice possibilities. The literature presents various criteria and indicators in fisheries management evaluation, the kinds and number of which would depend on stated goals and objectives of fisheries and the availability of resources to acquire the

M.N. Andalecio (🖂)

Institute of Fisheries Policy and Development Studies, College of Fisheries and Ocean Sciences, University of the Philippines Visayas, Miag-ao, Iloilo 5023, Philippines e-mail: merlina\_andalecio@yahoo.com

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_14,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Andalecio, Agron. Sustain. Dev. 30 (2010) 557–580. DOI: 10.1051/agro:2009051

information. The type of measurement, i.e., quantitative or qualitative, and the weighing of criteria and indicators are crucial in the evaluation process because they determine the multi-criteria aggregation approach to be used. Moreover, the participation of stakeholders and coastal resource users is crucial in complementing scientific information, in developing acceptable solutions, and in reducing conflicts and distrust in the evaluation and decision-making process. While many aggregation models in multi-criteria analysis in natural resource management exist, this article limits its review to only six models: the analytic hierarchy process, the weighted sum model, the ordination technique, concordance analysis, the regime method and Evamix; which are viewed to be applicable to the structure of decision-making in tropical fisheries management. This article also examines the performance of some of these models through a case study that determines the impacts of fisheries management strategies in San Miguel Bay, Philippines. The review reveals the following: (1) among the aggregation approaches, the analytic hierarchy process and ordination technique had the highest number of applications in fisheries while none was found for concordance analysis, the regime method or Evamix; (2) the application of hybrid models in multi-criteria analysis is increasing and found to be effective in many environmental decision problems including fisheries; (3) the application of multi-criteria decision models to fisheries management is relatively scarce during the last 10 years; only 26 papers were found in peer-reviewed journals; and (4) in the choice of model, its technical assumptions and limitations, its appropriateness for a specific decision-making problem, and its ability to handle the situation correctly vis-à-vis contextual, technical and political concerns should be considered.

**Keywords** Multi-criteria decision models • Fisheries management • Impact evaluation • Indicators • Criteria • Preference system • San Miguel Bay

### Abbreviations

AHP	Analytic hierarchy process
BFARMCs	Barangay fisheries and aquatic
	resource management councils
CADS_TOOL	Cage aquaculture decision support
	tool

CI	Consistency index
СР	Compromise programing
CR	Consistency ratio
EBM	Ecosystem-based management
ELECTRE	Elimination and choice translating
	reality
Evamix	Mixed data evaluation method
FARMCs	Fisheries and aquatic resource
	management councils
FSP	Fisheries sector program
GIS	Geographic information systems
LGP	Lexicographic goal programing
MCA	Multi-criteria analysis
MCDM	Multiple criteria decision-making
MCE	Multi-criteria evaluation
MDS	Multidimensional scaling
MOP	Multiple objective programing
MPA	Marine protected area
MPAEM	Marine protected areas evaluation
	model
MUF	Multi-attribute utility function
Rapfish	Rapid appraisal technique for
	fisheries
RI	Random index
SIC	Sustainability index of criteria
WGP	Weighted goal programing
WSM	Weighted sum model

### 1 Introduction

Fish is an important protein source for over 2.9 billion people, providing a livelihood to an estimated 43.5 million people worldwide (FAO, 2009). However, due to excessive fishing pressure, the FAO (2009) reported that 80 percent of the world's fish stocks are fully exploited or overexploited, yielding less than their maximum potential or catching near maximum limits. The collapse of major international fish stocks such as the northern cod fishery and other ground fisheries is well documented in the literature (e.g., Myers and Quinn, 2002; Neis, 2000; Sinclair et al., 1999). This collapse did not happen overnight (Jensen, 2002) but is a result of overfishing, long-term mismanagement, increased population growth, development and improvement of fishing technologies, uncertainty in global fisheries' catch data, economic incentives and subsidies, increasing demand for fish meal, etc.

(Zeller and Pauly, 2005; Eagle and Thompson, 2003; Jensen, 2002; Bundy and Pauly, 2001; McManus et al., 1997; Caddy et al., 1998).

The management of complex and heterogeneous tropical coastal fisheries is difficult. For decades, interrelated problems including overfishing, habitat degradation, natural resource depletion, widespread poverty in coastal communities, encroachment of commercial fishing vessels inshore, use of illegal and destructive fishing methods, resource use conflicts, siltation and pollution from uplands, weak institutional arrangement and many others have received extensive attention. These problems are especially apparent in many tropical countries given the multispecies and multigear nature of coastal fisheries. In response to these growing problems and in the hope of reversing their negative effects, several fisheries management interventions, collectively termed as fisheries management strategy, have emerged for coastal fisheries. Fisheries management interventions include establishment of marine protected areas, area closures, mesh size regulations, gear restrictions, by-catch restrictions and controls, fishing effort restrictions, limited entry licensing, comprehensive monitoring and enforcement, temporal and spatial distribution of fisheries, and alternative employment (Lowry et al., 2009; Crawford et al., 2004; Wilson et al., 2002; Zeller and Reinert, 2004; Witherell et al., 2000; Pollnac et al., 2001; Allison and Ellis, 2001). Most of these fisheries management interventions were implemented in many developing countries' coastal fisheries during at least the past 30 years.

The universal goal of sustainable fisheries management is to improve the state of the fisheries for the benefit and enjoyment of the present generation and generations to come. These goals are specified in Charles (1989) and Cochrane (2000) to include resource conservation, food production, generation of economic wealth, generation of reasonable incomes for fishers, maintenance of employment for fishers, maintenance of the well-being and viability of fishing communities, sustainable utilization, economic efficiency, and equity in access to resources. These can be broadly categorized as ecological/biological, social, and economic objectives. Because of the diversity of the coastal fishery resources and interest groups, compromises in goals and objectives are often necessary (Mardle et al., 1997) because it is impossible to maximize different objectives simultaneously (Munda, 2005). This is the case in tropical developing countries

where national goals and objectives of fisheries sometimes may not complement the needs of the local community. For example, conflicts often occur among attempts to conserve fish stocks, maximize efficiency and export earnings, and the desire to satisfy the needs of the fishing communities with respect to jobs and income (Khorsid and Morgan, 1990; Mardle et al., 1997; Whitmarsh, 1998). Conservation or preservation of biological diversity is often contrary to the aim of many fisheries organizations in maximizing production or economic efficiency (Agardy, 2000). In addition, increasing one objective such as regional benefits tends to decrease other goals such as national efficiency (Sylvia and Cai, 1995).

In fisheries, it is crucial to determine the outcomes of management strategy, especially when public money spent on projects and programs has to be accounted for. The success or failure of a management strategy has to be dealt with in an integrated manner that accounts for the multiple objectives of the fisheries. If the need is to regenerate, rehabilitate, conserve, protect and sustainably manage the fisheries resources, then the management of the coastal areas should be based on the understanding of the interrelationships of the various parts of the ecosystem (Griffis and Kimball, 1996; Legendre and Legendre, 1983) including human interactions. Many program evaluation reports or documents have failed to do away with mere descriptive summary of the findings or present an analytical framework to assess the performance of the fisheries management strategy.

Becker (2001) defined impact assessment or evaluation as "the process of identifying the future consequences of a current or proposed action". Fundamental to any evaluation is the method or tool to be used. In fisheries management, the importance of developing analytical and operational evaluation tools is critical for sound decision-making. Evaluation of fisheries management is often atomistic in nature which means that impact evaluation is frequently undertaken through a single-disciplinary approach (either a biological or an economic approach). Biological impacts are determined by abundance, biomass and sizes of fish, species composition, and diversity (Dinmore et al., 2003), whereas economic impacts are evaluated through monetary measures such as benefitcost analysis, travel cost, hedonic pricing, inputoutput analysis, contingent valuation, etc. (Dalton, 2004; Hoehn, 1987; Propst and Gavrilis, 1987) as

well as non-monetary techniques such as social wellbeing, psychophysical measures and attitude measures (Dixon et al., 1994; Gregory, 1987). One limitation of each approach is that the impacts are usually considered uni-dimensional, as demonstrated in the studies of Heen (1989) and Karydis and Coccosis (1990).

Because of the limited ability of current scientific methods to measure and understand with certainty the ecological processes and the high temporal and spatial variability and uncertainty in the biological components of the marine environment (Parr et al., 2003), it may be insufficient to consider only the biological impacts of management. What if scientific methods fail to detect the impacts on their underlying causality: does it mean that management has no effect at all? For example, McManus (1986) reported that in the Philippines, although a ban on commercial trawling had gradually been imposed on the fishery over several years, the degree of impact of the ban on the local species composition was not known. Many of the costs and benefits of management are difficult to quantify, and even if quantifiable, may be difficult to measure in monetary terms (Bingham et al., 1995). In addition, the social and institutional impacts of management (especially the distributional aspects) are not explicitly included in either biological or economic impact assessments. The social aspect is concerned with the survival of coastal communities dependent on the fisheries and can be evaluated as community participation and cooperation, employment, change in the degree of user conflicts, improved standard of living, etc. The institutional one deals with governance or the administrative and political aspects of management.

When the intention of the evaluation is to examine the multiple effects of management strategy, a single approach may no longer provide sufficient estimates of impacts because it precludes a meaningful evaluation of the complexities of fisheries and factor interaction. The current direction therefore is to consider the multidimensionality of fisheries (i.e., biological/ecological, economic, social and institutional). In which case, a multiple criteria (or multi-criteria) type of evaluation is a potentially useful analytical tool that can complement (Nijkamp et al., 1990) [not compete with or replace] and strengthen the existing approaches. Unlike any of the single approaches referred to earlier, multi-criteria methods deal with mixed information measured on varying scales either quantitative (also known as *cardinal* – i.e. interval and ratio scales) or qualitative (i.e., ordinal or nominal/categorical scales) or both.

The purpose of this paper is to review the extent of the application of multi-criteria evaluation methods to tropical fisheries management. This paper is divided into sections. Section 1 introduces the problems and challenges related to the discipline-based evaluation of the impacts of management in tropical fisheries, and a rationale for the use of a multi-criteria type of evaluation. Section 2 presents the framework, structure and features of a multi-criteria evaluation method, with extensive discussion on the selection, measurement and weighting of criteria and indicators as well as stakeholders' participation. Section 3 reviews and examines some aggregation approaches usually applied in natural resource management and that show potential in tropical fisheries management. Section 4 presents a case study in fisheries impact evaluation and provides information on the number of applications of multi-criteria methods to fisheries management. The paper closes with a discussion on the limitations of multi-criteria methods and the need for a systematic evaluation in tropical fisheries management.

# 2 Fundamentals of Multi-Criteria Evaluation Method

Multiple criteria evaluation approaches may appear in the literature as multi-criteria analysis (MCA), multiple criteria decision-making (MCDM) or multi-criteria evaluation (MCE). Regardless of nomenclature, the intention is to examine a number of choice possibilities (e.g., alternative plans or strategies, management options, administrative zones or regions, etc.) through multiple criteria that measure the attainment of conflicting goals or objectives with the explicit inclusion of subjective weights. This allows different points of view to be identified and explored (Chesson et al., 1999; Mahmoud and Garcia, 2000; Van Delft and Nijkamp, 1977). Multi-criteria analysis appeared in the 1960s and 1970s as a decision analysis tool in response to the growing operations research in WWII (Hajkowicz and Collins, 2007). They have demonstrated their utility in many environmental issues that link economic, environmental, cultural and technical issues of management (Abu-Taleb, 2000; Makowski et al., 1996). The advantages of multicriteria evaluation models are explicitly discussed in Hajkowicz (2008), Bryan and Crossman (2008), Hajkowicz and Collins (2007), Linkov et al. (2006), El-Gayar and Leung (2001), Brody et al. (2006), and Chesson et al. (1999) and are summarized as follows:

- useful in improving the decision-making process, leading to public acceptance of suggested options;
- applied in multi-stakeholder decisions providing a transparent, structured, rigorous and objective evaluation of options;
- dispute resolution tool that assesses progress of a particular objective by reducing the level of complexity of information;
- help balance conflicting demands of environmental conservation and development with regards to water allocation and coastal development.

The general structure of a multi-criteria evaluation method consists of at least a two-dimensional matrix, where one dimension expresses the various choice possibilities while the other dimension is composed of criteria that will evaluate these choice possibilities (Voogd, 1983). Munda et al. (1994) and Nijkamp et al. (1990) summarized the procedure as follows:

- (a) structuring the problem (defines what is to be evaluated);
- (b) generation and definition of choice possibilities (e.g., alternatives);
- (c) choice of a set of evaluation criteria;
- (d) determination of scores for each criterion and indicator per choice possibility (the value assigned to the choice possibility with respect to a criterion);
- (e) identification of a preference system of decisionmakers (criterion priorities have to be defined so that information can be combined); and
- (f) selection of an aggregation procedure (choice of methods to come up with an analysis of the evaluation problem).

The outcome of the above procedure from (a) to (d) is an evaluation matrix composed of a list of criteria and the scores of the criteria for the choice possibilities. A number of existing multi-criteria methods have a similar structure of the evaluation matrix. Their only difference would be on how steps (d), (e) and (f) in the procedure will be handled. Prior to the discussion on the aggregation procedure, the primary components of the evaluation matrix in relation to fisheries management are presented below.

# 2.1 Choice of Evaluation Criteria and Indicators

The importance of developing evaluation criteria and performance indicators (structural and functional elements used to judge the success of management) for project appraisal (Van Pelt, 1993; Ware et al., 2009), habitat restoration (Pastorok et al., 1997), management programs (Anderson, 1989), ecosystem approach (Jennings, 2005), and sustainability assessment and management (Froese, 2004; Ehler, 2003; Garcia et al., 2000; Hardi et al., 1997) is well-recognized. The criteria and related indicators are often derived from goals and objectives of fisheries management and according to Bonzon (2000), the government and local authorities are likely to select them based on their own specific objectives.

The terms criteria and indicators are often used interchangeably in the literature. A criterion may be defined as a concept designed to specify the expected or desired outcomes of implementing a management strategy. It may not be an actual measure since there is no single exact measure for any of the criteria. Instead, one or more indicators or performance indicators measure a criterion. The kinds and number of criteria selected largely depend on the stated goals and objectives. When goals and objectives of fisheries are vaguely defined, more effort is needed to obtain specificity of the criteria; although sometimes vagueness of goals and objectives is inevitable to allow flexibility for changing program activities as future circumstances necessitate (Rutman, 1984).

Most criteria associated with fisheries sustainability and management that appear in the literature include feasibility (Yahaya, 1988; Yap, 2000), economic efficiency and benefit (Anderson, 1989; Bonzon, 2000; Tam et al., 1996; Whitmarsh, 1998), equity (Bonzon, 2000; Nickerson, 1999; Van Pelt, 1993; Yahaya, 1988), acceptability (Yahaya, 1988), ecosystem health and integrity (Linton and Warner, 2003; Wells, 2003), social welfare (Tam et al., 1996), effectiveness (Sumaila et al., 2000), enforceability of the management program (Maliao et al., 2004; Anderson, 1989; Yahaya, 1988), ecological accountability (Reynolds, 1993), institutional performance (Imperial, 1999), and biological diversity (Mawdsley and O'Malley, 2009). Each criterion may also be categorized under a broader classification. For example, Sutinen (1999) referred to biological, social, economic or administrative categories.

Concerns have been raised as to the number of criteria. If too many are used the process becomes unmanageable, or if too few, the evaluation process may become oversimplified. But how many are too many or too few, in fact, depends upon the availability of administrative resources or logistics to acquire the information. Even if an enormous number of criteria that incorporate biological, social, economic and administrative dimensions of managing a tropical fishery are identified, the choice still depends on whether they are "policy relevant, scientifically reliable and valid, simple, sensitive, possible to aggregate, affordable and feasible in terms of data collection" (Hanson, 2003). When baseline information with which to compare the current data is incomplete, existing documents that indicate standards, thresholds or reference points may be used. In a developing country, the validity of an evaluation is often challenged because baseline information is frequently insufficient and methods of data collection are inadequate to allow comparability of results (Pomeroy et al., 1997). The problem may not actually be incompleteness of data, but whether available data are reliable or not; and how to delineate those which are perceived useful to ensure a meaningful evaluation.

The sources of information may not only be those which were acquired through scientific means. If the knowledge base that we have is all that is available (Lane and Stephenson, 1995) then, Johannes' (1998) strong argument for a data-less management - "that is, management carried out in the absence of the data required for the parameterization and verification of models that predict effects of various management actions with useful confidence limits" justifies the inclusion of local knowledge (e.g., fishers' knowledge of the coastal waters and resources) in the evaluation process. He emphasized that management is not to be judged by its roots but by its fruits. Studies such as those of Berkes et al. (2001), Salas and Gaertner (2004), Ticheler et al. (1998), and White and Savina (1987) support the successful participation of fishers and local stakeholders in research, data collection and use of traditional knowledge in the formulation of scientific hypotheses. Thus, there should be no reason to question the participation of fishers in the evaluation process.

An indicator is a specific and straightforward measure. Bonzon (2000) characterized indicators as tools for measurements as value variables (either quantitative or qualitative), indices or pointers related to criteria of a given system. They are single measures of a resource element in an unaggregated form (Knuth and Nielsen, 1989) used to describe the state of the system and assess trends (Garcia et al., 2000). An indicator is not only a measurement tool, it is also a way of defining what is measured (Harte and Lonergan, 1995) and how will it best serve the users' needs (Rice, 2003). In general, indicators may be specific to particular uses and use contexts in both scale and content (Dahl, 2000); thus, they need not be generalized. Sustainability indicators are commonly categorized into discipline (e.g., indicators on biological, economic, social or cultural aspects), or according to whether they measure the factors that exert pressure, show the present system state or indicate responses of concerned groups to system changes or the pressure-stateresponse model (Adriaenssens et al., 2004; Willmann, 2000). Hundloe (2000) cautioned the use of an indicator that is only based on an economic measure because economy is only part of the environment and human system. Frequently employed ecological, social and economic indicators in fisheries are discussed in Garcia et al. (2000), Garcia and Staples, (2000) Kabuta and Laane (2003), Gislason et al. (2000), Pastorok et al. (1997), Staples (1997), and Vandermeulen (1998) among others.

Comprehensive listing of attributes and considerations in developing indicators are found in Ehler (2003), Kabuta and Laane (2003), Chong (2000), Vandermeulen (1998), Cairns et al. (1993), and Pastorok et al. (1997). These include responsive to change, supported by reliable and readily available data, relevant to the issue, unambiguously expressed, have broad relationships with existing evaluation tools, scientifically valid, national perspective, cost-effective, and if possible, predictive, easily understandable, simple and acceptable to intended users. The last attribute strengthened Staples' (1997) earlier point about the importance of considering the main users of the indicators in impact evaluation. Various resource users or decision-makers may make different choices of indicators. For example, scientists or people from academia would prefer indicators that are basic such as water quality parameters while fishers or local government officials would opt for applied indicators such as income, number of boats, etc. According to Bonzon (2000), indicators can reflect the needs of various entities and stakeholders (e.g., management authorities, producer associations, or the general public).

He further stated that in selecting indicators, information needed mainly for academic research must be distinguished from information directly related to strategic management planning. The choice of the number of indicators may also depend on the group of decision-makers; politicians may prefer a few simple indicators while technical experts would more likely include large numbers of indicators (Dahl, 2000). Indicators should be able to reduce the number of individual variables and data points while maintaining a sufficient level of understanding about coastal systems (Bowen and Riley, 2003). However, there are technical and methodological complexities that come with aggregation and weighting of mixed indicators (Dahl, 2000).

While Staples' (1997) and Vandermeulen's (1998) basic premise that indicators should be able to compare information with a standard, target, threshold or limit value has merit, this is often difficult to satisfy since standards and thresholds are not always established or available for many indicators. If this is the case, then the indicators can be classified as benefit or cost indicators. This means that the higher the value of the benefit indicators, the more they are preferred. Similarly, the lower values of cost indicators are more preferred. For example, the abundance of commercial fish catch is considered as a benefit indicator.

The amount of information generated for the criteria may also limit the number of indicators that can be measured to only a fraction of those possible (Cairns et al., 1993; Propst and Gavrilis 1987). Often, a criterion contains two or more indicators. Since many fisheries criteria and indicators are interrelated or interdependent, the choice of evaluation methods has to consider interdependence. Some evaluation methods have very limited assumptions; for instance, there are those which would only allow analysis if the criteria or indicators are independent of each other. However, it is less likely to find fisheries indicators that are unrelated, especially those belonging to the same criterion. For example, the indicators number of trainings and seminars conducted and level of awareness of resource users intended to measure the criterion acceptability of management may somehow be directly related. Our knowledge of which indicator would best measure a criterion is quite limited. Our assumption is that an infinite number of indicators contribute to the measure of a criterion and the probability of an overlap, redundancy or double counting of the indicators is bound to exist because of the inherent interrelationships among them.

While the development of performance criteria and indicators for fisheries management requires technical information, Chong, (2000) emphasized that the interest, willingness and commitment of the people or the community are also imperative in the sustainable management and conservation of fisheries and other coastal resources. Therefore, these same people have to be part of the development of the performance criteria and indicators for sustainable management.

### 2.2 Measurement of Criteria and Indicators

Deriving the values of the indicators, and subsequently, the criteria is critical in the final evaluation process. Usually both qualitative and quantitative information associated with several criteria need to be systematically considered when evaluating several decision alternatives (Wenger and Rong, 1987). Indicators as measures of criteria are categorized according to the four types of measurement scales, namely, nominal, ordinal, interval and ratio scales. Van Delft and Nijkamp (1977) and Vogt (1999) characterized these scales as follows: in the nominal scale the numerical operations are pointless because the numbers only represent names having no order or value while in the ordinal scale, the subjects are ranked in an order such that differences between rank orders have meaning. The interval scale does not have a fixed origin but it allows some numerical operations such as averaging, addition or subtraction. The ratio scale has a true zero point; thus, all standard operations can be carried out on this scale. Interval and ratio data can also be collectively called 'cardinal data'. Data measured on an interval or ratio scale are either continuous or discrete (discontinuous); continuous data are placed in a scale with an infinite range of points while discrete data are made up of distinct and separate units or categories (Vogt, 1999; Wheater and Cook, 2000).

In fisheries, a criterion is measured quantitatively or qualitatively using two or more indicators. But even if all indicators are measured using only one type of scale (e.g., quantitative), the units of measures may not be homogenous (e.g, hectares, currency, tons, percentages, etc.). This is a type of scaling problem that was resolved through transformation of values into a common order of 0 to 1 (Yakowitz, 1998) or normalization to obtain comparable scales because each function may have a different number of variables or mathematical relationships and corresponding maximum scores (Hruby, 1999). Although there are different kinds of normalization formula, the most commonly used are found in Pomerol and Barba-Romero (2000), Voogd (1983) and Hwang and Yoon (1981) and these are presented below:

- (a)  $e_{ij} = x_{ij} / \Sigma x_{ij}$ where  $e_{ij}$  = normalized indicator score  $x_{ij}$  = score of the indicator  $\Sigma x_{ij}$  = sum of all indicator scores
- (b)  $e_{ij} = x_{ij}/x_{ij}^{max}$ where  $x_{ij}^{max} = maximum$  indicator score
- (c) vector normalization:
- $e_{ij} = x_{ij} / \sqrt{\sum} x_{ij}^2$ (d) linear scale transformation:
- (a) mean scale transformation:  $e_{ij} = (x_{ij} - x_{ij}^{min})/(x_{ij}^{max} - x_{ij}^{min})$ , for benefit criterion or,  $e_{ij} = (x_{ij}^{max} - x_{ij})/(x_{ij}^{max} - x_{ij}^{min})$ , for cost criterion where,  $x_{ij}^{min} =$  minimum indicator score;  $x_{ii}^{max} =$  maximum indicator score.

When using vector normalization, all indicators are measured in dimensionless units, thus facilitating interindicator comparisons, but direct comparison is difficult because the minimum and maximum values of the measurement scale are not equal for each indicator. Linear scale transformation has the advantage of transforming results in a linear (proportional) way, making the relative magnitude of the outcomes equal (Hwang and Yoon, 1981). Also, if the nature of the indicators is different, the indicators are grouped as benefit indicators (larger  $x_i$  or value of the indicator is more preferred) or cost indicators (smaller  $x_i$ or value of the indicator is more preferred). Some authors (e.g., Nijkamp et al., 1990) recommended that if normalization is done, it is best to test the sensitivity of the outcome for the particular type of normalization.

The level of measurement imposes special conditions on the techniques to be used in further data manipulation. Smith and Theberge (1987) presented three aspects of measurement theory that will be useful in understanding the measurement of the indicators: (i) four basic scales of measurements (nominal, ordinal, interval and ratio) define the types of mathematical operation to be applied to the values; (ii) measuring environmental variables or subjective values; and (iii) uncertainty in measurements which affect both how measurement is done and the confidence that is placed in the values obtained. Works on multi-criteria methods rarely discuss how the measures of indicators were arrived at – e.g., uncertainties attached to the measurement. Because of the temporal and spatial differences associated with the collection of information for each indicator [data for the indicators and criteria may not be collected at the same time], it would be useful to standardize the methods of data collection to develop a meaningful database or information system.

### 2.3 Weights of Importance of Criteria and Indicators

The importance placed on the criteria and indicators is another consideration-importance may be modeled statistically by means of rank orders (Yakowitz, 1998), rating scales, paired comparisons or magnitude estimates. Some studies used multiple regression analyses to predict judgmental values as a function of various physical features of the environment while others applied multivariate techniques such as factor and cluster analyses to learn more about interrelationships among the indicators (Gregory, 1987; Petry, 1990). Prato (1999) noted that while there is no theoretical limit to the number of criteria (which he referred to as attributes), an individual's ability to assign weights to these criteria decreases with their number. One method I found in the literature that is useful to address this issue is the Analytic Hierarchy Process (AHP), developed by Thomas Saaty in the early 1970s (DiNardo et al., 1989; Leung et al., 1998). The AHP is a method of scaling ratios using the principal eigenvector of a positive pairwise comparison matrix (Saaty, 1980, 2001). Although a multi-criteria evaluation tool, the AHP can be used to just derive the weights of importance of criteria and indicators and not directly proceed with the comparison of choice possibilities. It structures a problem into a hierarchy then the weights of the criteria (and also indicators) are determined through pairwise comparisons (Saaty, 1980) according to preference, importance or likelihood (Peterson et al., 1994). Through pairwise comparisons, evaluators or decision-makers (e.g., fishers, coastal resource users) will not be overwhelmed with the amount of information that has to be processed mentally per unit time. To assess the scale ratio, Saaty (1980) provided a 9-point intensity scale of importance. One important feature of the AHP is the determination of consistency in judgments. This is done by computing the consistency ratio (CR), which is the ratio of the consistency index (CI) to the average random index (RI). Thus, CR = CI/RI, where CI =  $(\lambda_{max} - n)/n - 1$  and RI is referred from a table of average random indices found in Saaty (1980). The  $\lambda_{max}$  is  $\Sigma$  (total normalized value of criterion  $z \div$  column vector of priorities of criterion z). The consistency ratio measures the coherence of the pairwise comparisons and estimates the level of consistency with respect to the entire comparison process. A CR of 10% or less is considered acceptable. The consistency ratio measure allows AHP users to be aware of the seriousness of any inconsistent judgments (Leung et al., 1998).

Although weighting of the criteria is open to criticisms (Munda et al., 1994) because it involves human judgment, it is indispensable in impact evaluation work. Petry (1990) emphasized that as all the simplifications and approximations necessary for scientific analysis have some human value content, technical analysis and political decisions cannot be completely separated.

# 2.4 Stakeholders and Resource Users as Evaluators

The determination of the importance of criteria and indicators would have been simple and easy had preference among individuals been known to be similar. This objectifies value judgment and thus, a single evaluator would then be sufficient to represent preference of a population. But utility functions vary between individuals and preferences vary over time. The problem with a subjective judgment is how to validate whether the derived weighting is representative of the judgment of the population. Bodini and Giavelli (1992) resolved this problem through a survey technique that isolated viewpoints from the subjectivity of the planner and facilitated the involvement of local communities in the decision phases. However, their work failed to account for conflicts that may ensue in the final analysis – if stakeholders are categorized into groups, will each group's perception vary from each other? When Leung et al. (1998) applied the AHP approach, geometric means of the judgment from respondents were used to derive the overall and the respective group's priority, then, variation among individual judgments within and between groups was determined using analysis of variance. Grouping the individuals according to their respective stake in the fisheries would be more rational than combining all individuals with varied interests because of the problems of divergence in preference and domineering behavior of some individuals. The process that Stewart and Scott (1995) identified to re-homogenize individual groups is too tedious because every time a consensus is not achieved within a homogenous group, either multiple points of view or sub-criteria are created, or the group is subdivided into two or more groups representing the different interests. Srdjevic (2007) emphasized that the characteristic and size of the group may be critical in the application of multi-criteria methods because of the assumption of homogeneity.

In a top-down approach type of management, there are only a few individuals whose views are likely to influence the final decisions of the management process. However, when larger and varied groups of individuals are involved, identification of a preference system has to be approached differently. With an increased number of participants (or evaluators), the computational burden increases (Prato, 1999); however, the problem of computational overburden need not concern the evaluators - this is the responsibility of the analysts (who most likely will use computer software in data analysis). The institutional framework in which the entire decision-making process occurs determines the categories and number of evaluators. An institutional structure that is systematic and flexible is able to facilitate the integration of a wide range of viewpoints through interaction between and among diverse groups of individuals with varying stakes in the fisheries (Petry, 1990; Ridgley and Rijsberman, 1994). Although technical information is important, stakeholders' knowledge, experience and judgments are crucial in the application of a multi-criteria evaluation tool. This is evident in a complex system such as fisheries where incomplete information and understandings may exist (Adrianto et al., 2005). For example, Chong (2000) recognized the invaluable contribution fisherfolk may have in providing feedback of the condition of the resources and habitats. Similarly, the importance of different types of outcomes of management options to different stakeholders is essential in developing an acceptable solution (Pascoe et al., 2009). Linkov et al. (2006) observed that failure to incorporate stakeholders into decisions resulted in distrust and political tension.

The way the multi-criteria evaluation method is structured allows for the involvement of stakeholders in the decision-making and evaluation process. However, despite the wide recognition of the crucial participation of fishers in the management of fisheries resources (Clarke et al., 2002; Castilla and Fernandez, 1998; Ferrer et al., 1996; Gilman, 1997), the mechanism through which to include them in the formal evaluation process is not actually established. There are challenges to this which include quantifying uncertainty related to human input and overcoming the difficulty in negotiation when a large number of groups are responsible in decision-making and prioritizing actions (Ascough II et al., 2008).

# 3 Multi-Criteria Aggregation Approaches

Following the structure of an evaluation matrix, we are now able to construct an "impact evaluation matrix", characterized as an ex post evaluation (Nijkamp et al., 1990; Voogd, 1983) which deals with the analysis of the effects of management strategies that were already implemented. The multi-criteria evaluation method is usually named after the kind of aggregation procedure applied in the analysis. It often uses two kinds of input data: criterion scores and a set of political weights attached to these criterion scores. When input data are completed, the final step concerns the aggregation procedure to determine the progress of a choice possibility, whether based on an already established standard or in comparison with other choice possibilities.

Extensive discussion of multi-criteria evaluation methods and aggregation approaches is beyond the scope of this paper. For recent reviews of the applications of multi-criteria methods, readers are referred to Hajkowicz and Collins (2007), and Linkov et al. (2006). Since it is impractical to conduct a thorough survey of all the different methods, I narrowed the review to the model-based decision support tools usually applied in natural resource management, particularly coastal management. These are the analytic hierarchy process, the weighted sum model (WSM), the ordination technique, concordance analysis, the regime method and the mixed data evaluation method (or Evamix). I view them as applicable to the structure of decision-making in tropical fisheries management; specifically, in addressing the issue of performance evaluation of management strategies.

### 3.1 Analytic Hierarchy Process (AHP)

In Section 2.3, the analytic hierarchy process was used to elicit the weights of importance of criteria and indicators to attain a set goal or objective. Conversely, the AHP has also been extensively applied as a multicriteria evaluation and decision support system in the last 20 years (Ho, 2008). The final result of the process is a numerical priority value for each choice possibility. The choice possibility with the highest score is considered the best one as determined by the decision process made explicit in the hierarchy and by the comparisons (Peterson et al., 1994). A more detailed analysis of the theoretical foundations and applications of the AHP appears in the papers of Vaidya and Kumar (2006), Saaty (2001), Triantaphyllou and Lin (1996), Ridgley and Rijsberman (1992) DiNardo et al. (1989), Khorramshahgol and Moustakis (1988), and Saaty (1980). The AHP uses the computer software expert choice to assist the evaluators in processing a large amount of information properly and performing sensitivity analysis.

The analytic hierarchy process has gained wide acceptance in the fields of urban and land development (Banai, 2005), water resource planning (Willet and Sharda, 1991), natural resource management and planning (Fernandes et al., 1999; Peterson et al., 1994; Schmoldt et al., 1994), restoration (Ridgley and Rijsberman, 1994), agriculture (Duke and Aull-Hyde, 2002), etc. The approach is highly dependent on the experience, knowledge and intuitive judgment of the evaluators. Ridgley and Rijsberman (1992) vouched for the use of this method as: (i) it does not demand independence of alternatives; (ii) it creates and is based on ratio scales rather than interval scales; (iii) it does not require that the range of criterion scores be known before comparison of choice possibilities relative to that criterion can be made; and (iv) it uses subjective assessments of preference intensity. These

advantages may be apparent if the intention is to apply the AHP beyond weighting of the criteria and indicators to prioritizing of choice possibilities. This was demonstrated in the studies of Dey and Ramcharan (2008), Herath (2004), Leung et al. (1998), Schmoldt et al. (1994), Peterson et al. (1994) and Ridgley and Rijsberman (1992). Leung et al. (1998) and Varis (1989) found the AHP to be effective and robust in solving large, complex and evasive decision problems. Successful application of the AHP is demonstrated in Ridgley and Rijsberman's (1992) policy analysis for a Rhine estuary and Peterson et al.'s (1994) resource management plans. However, it is only the study of Peterson et al. (1994) which took note of the actual length of time i.e., two days' discussion, to implement and accomplish the process. Although the authors did not encounter any problem in reaching a consensus, they suggested that if this happens, a separate judgment can be aggregated using a geometric average. Contrary to the works of Ridgley and Rijsberman (1992) and Peterson et al. (1994) [who used consensus building among the evaluators in order to arrive at a final solution using the AHP], Leung et al. (1998) [after realizing that the group of decision-makers is large and diverse] employed a mail survey type of instrument. The results of their study showed that a mail survey is not an efficient technique to administer the AHP because of the low response rate. That is, only 52% of the 66 members of the Western Pacific Fishery Management Council returned usable surveys. Although the authors attributed such a low response rate primarily to the geographical distance and non-fisheries background of some Council members, the problem may also have been the lack of a venue for interaction. Respondents who are unfamiliar with the AHP may have had difficulty understanding its application without interacting with people who are more knowledgeable about the approach. The AHP has also been used in combination with other decision-making tools such as has been documented by Srdjevic (2007), Vaidya and Kumar (2006), and Bojórquez-Tapia et al. (2005).

There are only a few studies which applied the AHP to fisheries. Vaidya and Kumar (2006) provided a comprehensive review of 150 papers on the different applications of the AHP, none of which relate to fisheries. In the review of Mardle and Pascoe (1999), they only found four applications of the AHP in fisheries. These are the management of Maryland's river herring fishery (DiNardo et al., 1989), management of Kenai river

Chinook salmon fishery (Merritt and Criddle, 1993), Eastern Finland's multispecies fishery (Kangas, 1995) and Hawaii pelagic fishery (Leung et al., 1998). Since the review of Mardle and Pascoe (1999), the present review was able to document 10 more applications of the AHP in fisheries management, summarized as follows:

- (a) Pascoe et al. (2009) utilized the AHP to determine the impacts of different management options by assessing the objectives of fisheries management. The method was applied to three Australian fisheries and the results showed that the benefits and costs of management options or measures vary across fisheries.
- (b) In developing a decision support tool for sustainable cage aquaculture, Halide et al. (2009) incorporated the AHP to select the best site for cage aquaculture. This was in combination with a decision support tool, CADS\_TOOL (cage aquaculture decision support tool).
- (c) Whitmarsh and Palmieri (2009) successfully applied the method to elicit the preferences of the public and stakeholders for the different objectives in salmon aquaculture. Their findings concurred with other research that social acceptability of aquaculture is closely linked to its perceived environmental impact. Although no remarks were made regarding the use of the AHP, their results demonstrated the importance of preferences of interest groups in influencing aquaculture policy.
- (d) When Utne (2008) evaluated the performance of the Norwegian cod-fishing fleet, she found the AHP to be a useful performance measurement tool that incorporates performance indicators (e.g., accident risk, employment, profitability, quality of the fish meat, catch capacity (technical), and greenhouse gas (GHG) emissions/acidification) into the problem hierarchy. The process provided a basis for examining the stakeholders' involvement in fisheries management decision-making.
- (e) Himes (2007) used the AHP to investigate preferences of stakeholder groups for performance indicators in evaluating marine protected area (MPA) management. They found the method to be strong, providing quantitative information on stakeholders' diverse attitudes and perceptions regarding MPA. It is also helpful in generating options for improved management and the possibility that stakeholders will accept and support the results.

- (f) In the study of Ramos et al. (2006), reef diving choices and preferences were calculated using the AHP. This assisted the decision-makers to identify the most relevant criteria for diving i.e., the hope of seeing unusual aquatic organisms, and the benefits of updating diving skills.
- (g) Nielsen and Mathiesen (2006) found the AHP to be a useful support tool for uncovering conflicts and eliciting preferences of various stakeholders involved in the management of the sand eel and Norway pout fisheries. The authors are optimistic that the preference process may improve accountability, openness and participation, and may contribute to good governance of said fisheries.
- (h) Mardle et al. (2004) examined the utility of the AHP in measuring the importance of management objectives in achieving sustainable management in key interest groups of the UK English channel fisheries. Through the AHP, qualitative and quantitative criteria were incorporated into the framework of analysis. What is interesting is the usefulness of employing a personal survey (as opposed to a mail survey), which allows important interaction between the participants.
- (i) Soma (2003) tested the performance of the AHP in the shrimp fishery sector in Trinidad and Tobago and found the method to be an "empowering, educating, focusing, facilitating and quantifying tool" in fisheries management. She concluded that the method is particularly useful in developing countries which are deficient in reliable quantitative data.
- (j) Fernandes et al. (1999) evaluated coral reef management options and found that the AHP had been successfully utilized to make 'good decisions' consistent with the community's desires to preserve the social, ecological and economic benefits of park management.

#### 3.2 Weighted Sum Model

The weighted sum model (also known as simple additive weightings) is the simplest and most commonly used method when all criteria are measured on cardinal scales, expressed in comparable units, and weights are assigned per criterion. This method is discussed in detail in Hwang and Yoon (1981), Nijkamp et al. (1990), and Triantaphyllou and Lin (1996). The criterion scores are standardized or normalized to be comparable and these normalized criterion scores are multiplied by their respective weights. The products are called the weighted scores and they are summed up over all criteria yielding a total weighted score or priority score for each choice possibility (Smith and Theberge, 1987). The choice possibility with the highest priority score is the one that performed well and is said to be the best choice possibility (P\*). The best choice possibility (P\*) is determined through the mathematical expression,

$$P^* = \max_{M \ge i \ge 1} \sum_{i=1}^{N} x_{ij} w_j, \text{ where } P^* \text{ is the priority score}$$
  
best of the choice possibility,

 $x_{ij}$  is the measure of performance of the ith choice possibility in terms of the jth criterion, and  $w_j$  is the weight of importance (Triantaphyllou and Lin, 1996). Usually the weights are normalized so that

$$\sum_{J=1}^n w_j = 1.$$

The method requires that criteria scores are both numerical (i.e., interval or ratio scales) and comparable because the regular arithmetical multiplication and addition measures are employed (Hwang and Yoon, 1981; Hajkowicz, 2008). The scores of the criteria have to be comparable because of the process of combining attributes such that a 'high' score for one criterion must receive about the same numerical values as 'high' scores of other criteria. There is a drawback in this kind of approach – it is difficult to interpret the multiplication of criterion values by weights. Consider, for example, criteria X and Y; the score of criterion X (0.8) multiplied by its weight (0.1) would yield the same product as that of criterion Y with a score and weight of 0.4 and 0.2, respectively. The problem lies with the tendency for the criteria to offset each other. Being based on additive utility assumptions, the WSM assumes transitivity of preferences and comparability of any pair of actions. Thus, it is more applicable to single-dimensional (all units of measurements are similar) than multidimensional (units of measurements are different) problems. Also, this method considers independence of criteria and indicators, when in reality they are complementary (excellence with regard to one

criterion enhances the utility excellence with regard to the other criteria). It is a powerful tool as long as no important complementarities exist among the criteria (Hwang and Yoon, 1981). In the evaluation of fisheries management, the data available [especially ecological data] are likely to violate this assumption; therefore, the WSM may be of little use. Further, since all criteria are aggregated to obtain a single final result, the technique jeopardizes intermediate analysis (Petry, 1990; Triantaphyllou and Lin, 1996). According to Voogd (1982), one limitation of the WSM is that the outcomes will strongly depend on the (usually arbitrary) choice of the origins of the various measurement scales used. Chesson et al. (1999) applied the WSM technique to estimate the performance of the south-east fishery over time (i.e., 1993, 1994 and 1995) and examine the effects of fishing on two components, i.e. humans and the environment. In each component, quantitative indicators were identified. The weights of the components and indicators were determined from three groups of individuals: (1) individuals who value both economic and ecological components, (2) those who value short-term financial returns, and (3) those who reflect conservation preference. The output of the analysis is a trend over time showing the progress of said fishery towards ecologically sustainable development. One important aspect of the process is the aggregation of weighted scores of the components. The authors, however, noted that the level of aggregation can be used to reduce the information to manageable amounts without being misleading. When the WSM model was used together with the Borda Count method, Hajkowicz (2008) found that it helped stakeholders make group decisions despite strongly conflicting preferences. When the assumptions in the application of the WSM are ignored, Hajkowicz and Higgins (2008) presented two valid issues: (a) the results showing only very minor differences in utility may be insufficient to differentiate performance, and (b) incorrectly treating ordinal data as cardinal data.

### 3.3 Ordination Technique

Identified as a multivariate method, ordination techniques in ecological research are used to quantify the interrelationships among a large number of interdependent variables and to explain those variables in terms of a smaller set of underlying dimensions (e.g., components) (McGarigal et al., 2000). The type of ordination technique that is commonly used in a multiple criteria analysis is the geometric or multidimensional scaling (MDS) ordination technique. It is a qualitative evaluation approach based on the ideal point concept (Voogd, 1980) and where quantitative inferences can be drawn without violating the ordinal character of the input data (Voogd, 1982). However, when no ideal point is identified, the minimum and maximum criterion values may be used (Hajkowicz and Collins, 2007)

Multidimensional scaling is very useful when dealing with too many criteria, which in some cases are vague and unknown. It uses proximities of pairs of choice possibilities in constructing a multidimensional spatial representation. When the information is on a ratio scale, it is converted to rank order dis(similarities). Hwang and Yoon (1981) and Stalans (1995) described the detailed operation of non-metric MDS as an approach that looks for a configuration and this configuration gives the spatial representation of the patterns of proximities among objects. Choice possibilities or alternatives are represented as points in space where point distances correspond to rank order of dissimilarity judgments among choice possibilities. Points near each other are assumed to be close together in terms of preference. The evaluator locates his ideal point and the distance from the ideal point is measured (using Euclidean distance or another measure) in order to rank the choice possibilities in terms of preference. Interpretation of the relative positions of choice possibilities in space is associated with the characteristics of the criteria that were scaled. One way to determine which among the criteria contribute to the positioning in the configuration is through multiple linear regression - with the criterion as the dependent variable and the coordinates of the configuration as the independent variables. This would mean that the criteria are regressed over the coordinates of the configuration. The distance measure used to form the configuration assumes that the criteria are independent or non-complementary. The scores of the criteria may take any form since the scaling procedure produces numerical and comparable values of each resultant dimension. None of the dimensions correspond with the single criterion of the original matrix. This attribute is particularly useful when the number of criteria is large (around seven) and most criteria are expressed qualitatively (Hwang and Yoon,

1981). This approach, however, requires that the evaluation problem has enough degrees of freedom (Voogd, 1982).

Although not used directly as a multi-criteria method, non-metric MDS was useful in the study of Moriki and Karydis (1994) as a multivariate statistical tool to support the results of multi-criteria methods (i.e., concordance analysis, the numerical interpretation method, and the regime method). Multi-criteria methods yielded similar ranking for sampling sites along the coastal areas of Rhodes and Saronicos Gulf (Greece) according to nutrient and chlorophyll concentration. Three levels of nutrient loading characterized eutrophic, mesotrophic and oligotrophic waters. Nonmetric MDS was applied to obtain a graphical representation of the ranked sampling sites. Moriki and Karydis (1994) only considered ecological indicator data. However, in the study of Moriki et al. (1996), ecological variables were integrated with socio-economic variables. They presented a case study of the island of Rhodes, Greece, which simultaneously analyzed economic, social and ecological data of the coastal system. The island was divided into five zones and the analysis was focused on the socio-economic characteristics of the zones, and chemical and biological parameters of the marine environment. Non-parametric MDS was one of the methods used to analyze coastal use conflicts, explore the development potential of the island and develop guidelines for future planning and decision-making. Similar to the other methods, MDS results showed a clear difference between intensity of land use and environmental quality, resulting in main aggregates, i.e., variables characterizing the quality of natural resources and the others characterizing the intensity of land use.

The ordination technique, MDS in particular, was applied in the development of Rapfish (rapid appraisal technique for fisheries), a multi-disciplinary rapid appraisal technique for evaluating the sustainability of fisheries with respect to ecological, economic, social, technological and ethical fields (see Pitcher, 1999). The technique constructs the best and worst possible fisheries from sets of scored attributes that were derived from fixed reference points (ideal points) or random reference points. MDS was able to generate ordination scores that provided a rating for a fishery from 0% (bad) to 100% (good). Rapfish has been applied in evaluating the status of fisheries globally such as 32 African lake fisheries (Preikshot et al., 1998), 42 Canadian fisheries (Pitcher, 1999), sustainability of fisheries in the North Atlantic (Alder et al., 2000), small pelagic fisheries (Pitcher and Preikshot, 2001), seven Tagus estuary fisheries (Baeta et al., 2005), 26 red sea fisheries (Tesfamichael and Pitcher, 2006), Basque trawl fisheries (Murillas et al., 2008) and ecosystem-based management (EBM) of fisheries in 33 countries (Pitcher et al., 2009). The marine protected areas evaluation model (MPAEM) (Alder et al., 2002) is another method which is based on the principle of Rapfish that explored the multidisciplinary approach of evaluating a single management intervention. Although both methods use multidimensional scaling as the analytical tool of the techniques, Rapfish evaluates the sustainability of fisheries while the MPAEM assesses the effectiveness of the management of existing MPAs. Six evaluation fields were defined in the MPAEM; namely, living (renewable) resources, non-living (non-renewable) resources, economic (market value), social, ecosystem functions and management (Alder et al., 2002). The model was tested in 20 MPAs in tropical, subtropical and temperate countries. While revisions of the model are imperative, it has shown potential for evaluating MPA management based on the evaluators' response. The most crucial part of the methods is the assignment of scores of the criteria per choice possibility which should be done by experts and well-experienced managers, researchers and stakeholders.

#### 3.4 Concordance Analysis

Concordance analysis, also known as ELECTRE (eliminating and choice translating reality), is based on pairwise comparisons of several choice possibilities. It was originally introduced by Benayoun et al. in 1966 (Hwang and Yoon, 1981), and since then, the method has been modified in the works of Van Delft and Nijkamp (1977), Roy (1991), and Pomerol and Barba-Romero (2000). The method makes use of successive assessments producing ranking of choice possibilities rather than indicating the most preferred (Chung and Lee, 2009). It is an outranking technique where choice possibility that is dominated or outranked the most is eliminated. The important input is a set of weights, and the output is a set of outranking relationships (or partial orders). Scores that form the impact

matrix are used to pairwise compare choice possibilities (Bodini and Giavelli, 1992).

Compared with the weighted sum model, concordance analysis is not based on utility theory. Utility functions are not used because of the substitutability property of the method where a bad outcome for a certain criterion can be compensated for by a good outcome for the other criteria (Hwang and Yoon, 1981). It is an evaluation method that can be used for cardinal data in the impact matrix and the weights vector. It examines both the degree to which the preference weights are in agreement with pairwise dominance relationships and the degree to which weighted evaluations differ from each other. The strength of this method is that it is able to underscore the two most important components of an evaluation matrix- the weighting and scores of the criteria. No assumption was made that the importance of the criteria is implicit in the evaluation scores; rather, weighting is explicitly considered. This method, however, assumes that the weighting and scores are derived quantitatively; although ordinal scores can also be used. Because the method clearly assumes that scores are cardinal in nature, the ordinal scores may be cardinalized to satisfy this assumption. This cardinalization of ordinal criteria or indicators is an indirect approach of transforming qualitative information into quantitative, and this is especially useful in mixed types of evaluation (Nijkamp and Vindigni, 1999). Then, all indicator scores are normalized to be comparable. One disadvantage of concordance analysis is the use of threshold values that can be selected arbitrarily. Because of this limitation, Van Delft and Nijkamp (1977) in Hwang and Yoon (1981) introduced the concept of net dominance relationships to complement the analysis. Two components of net dominance were presented net concordance dominance value and net discordance dominance value. Elements in the concordance and discordance matrices are used to calculate the net concordance and net discordance values, respectively. The central point of the method is the determination of the concordance and discordance indices (Moriki and Karydis, 1994). Salient elements and detailed calculations of the method are found in Chung and Lee (2009), Hwang and Yoon (1981), and Nijkamp et al. (1990). Keeney and Wood (1977) remarked that with ELECTRE it is difficult to do sensitivity

analyses to see just how much better one system is than another, but the latest version (i.e. ELECTRE IV) is able to incorporate sensitivity analysis. Moriki and Karydis (1994) found concordance analysis useful in identifying distinct areas of pollution according to nutrient characteristics of coastal waters.

#### 3.5 Regime Method

The regime method is a generalized form of concordance analysis based on a pairwise comparison method whose point of departure is an ordinal evaluation matrix and an ordinal weight vector (Nijkamp et al., 1990; Nijkamp and Torrieri, 2000; Vreeker and Nijkamp, 2001; Hinloopen et al., 2004). Nijkamp and Torrieri (2000) discussed its operation and summarized as follows: A concordance index  $(C_{ii'})$  is computed which is the sum of the weights of the criteria/indicators for which choice possibility i outperforms i'. The same procedure is also done when comparing choice possibility i' with i ( $C_{i'i}$ ). The difference between  $C_{ii'}$  and  $C_{i'i}$  yields the value of the index. Due to the ordinal nature of the information contained in the evaluation matrix, the magnitude of the difference between choice possibilities is disregarded. Unlike in concordance analysis, whose focal point is the concordance index, in the regime method, it is the sign of the difference for each pair of choice possibilities (Moriki and Karydis, 1994). The numerical size of the difference of the indicator for each pair of comparisons is ignored (Nijkamp and Torrieri, 2000). Pairwise comparisons are recorded in a table called a regime matrix composed of Z(Z-1) comparisons where Z is the number of choice possibilities. Then, a net concordance dominance index  $(\mu_{ii'})$  is calculated, where  $\mu_{ii'} = c_{ii'} - c_{i'i}$ . According to Nijkamp et al. (1990), the analysis aims to avoid the difficulty of having ambiguous results by partitioning the set of feasible weights, so that for each subset of weights a definite conclusion can be drawn about the sign of the index  $\mu_{ii'}$ . Because of the ordinal nature of  $\mu_{ii'}$ , its size is not the focus of the analysis but the sign. A positive (+) value of  $\mu_{ii'}$  means that choice possibility i is preferred over i' and negative (-) when the reverse is true (Moriki et al., 1996; Moriki and Karydis, 1994).

#### **3.6 Mixed Data Evaluation (or Evamix)**

A clear disadvantage of the qualitative methods (e.g., regime analysis) is that the available quantitative information is partially used (only the ordinal rank characteristics). Although the ordination technique can also be used to analyze mixed data, a better set of methods has been developed to deal with mixed qualitative-quantitative evaluation scores. Just like Concordance Analysis and the Regime Method, Mixed Data Evaluation, commonly called *Evamix*, is classified as a relative multi-criteria evaluation because there is no ideal value and the final appraisal score does not provide the absolute quality of a choice possibility; it only shows how different a certain choice possibility is with respect to the others (Voogd, 1982).

Evamix is based on the principle of analyzing information which has both quantitative and qualitative properties. It is also a generalized form of concordance analysis except that separate indices are constructed for qualitative (ordinal) criteria O and quantitative (cardinal) criteria C (Martel and Matarazzo, 2005). The difference between two choice possibilities can be expressed in a condensed way by means of two dominance measures, i.e. the ordinal dominance score  $(o_{ii'})$ and cardinal dominance score.

Ordinal dominance score  $(o_{ii'})$ 

$$o_{ii'} = \left\{ \sum_{j \in O} [w_j \bullet \text{sgn}(e_{ji} - e_{ji'})]^{\gamma} \right\}^{1/\gamma}$$

Cardinal dominance score  $(c_{ii'})$ 

$$c_{ii'} = \left\{ \sum_{j \in C} [w_j | e_{ji} - e_{ji'} |]^{\gamma} \right\}^{1/\gamma}$$

where

 $e_{ji} = \text{ score of criterion } j$  and choice possibility i

 $w_i =$  weight attached to criterion j.

$$\begin{split} \text{sgn}(e_{ji}-e_{ji'}) = +, \ \text{if} \ e_{ji} > e_{ji'} \\ 0, \ \text{if} \ e_{ji} = e_{ji'} \\ -, \ \text{if} \ e_{ji} < e_{ji'} \end{split}$$

 $\gamma$  = denotes an arbitrary scaling parameter which in this case will be assigned a value of 1.

In the determination of  $o_{ii'}$  only the ordinal characteristics of  $e_{ii}$  variables are considered, while for  $c_{ii'}$  the metric properties are also considered. Both measures are standardized to be comparable. Then, the total dominance score is computed as the weighted sum of the cardinal and ordinance dominance scores (Chung and Lee, 2009). The procedural framework of Evamix uses three approaches in order to come up with an appraisal score: the subtractive summation technique, subtracted shifted interval technique and additive interval technique. Detailed computation and application of the Evamix procedure are found among others in Voogd (1982, 1983), Nijkamp et al. (1990), Martel and Matarazzo (2005), Hajkowicz and Higgins (2008). However, Hinloopen et al. (2004) pointed out that the disadvantage of Evamix is that the separation of cardinal and ordinal information into two groups introduces a utility tree that is not based upon the preference structure of the decision-maker but on the level of information of the criteria. While published applications are rare compared with other multi-criteria analysis techniques (Hajkowicz and Higgins, 2008), the concepts and principles of Evamix are applicable to the structure of the problems in tropical fisheries management.

# 3.7 Summary of Multi-Criteria Aggregation Approaches

A vast and diverse number of aggregation techniques in multi-criteria analysis exist for different environmental decision-making problems. The choice of technique may actually depend on the nature and structure of the problem and information. It is important to make sure that the model is appropriate to the decision problem and complements the available data and not the other way around, i.e. forcing the data to fit the model. Thus, the limitations and assumptions of each model should be considered (see Table 1). The weighted sum model, analytic hierarchy process and Electre approaches rely on the provision of relative weightings of the criteria by stakeholders (Ascough II et al., 2008). When criteria or indicator scores and weights are both determined on a cardinal scale, then the simple weighted sum model is applied. This method, however, assumes that criteria/indicators are independent of each other, a condition unlikely to be satisfied in fisheries assessment as many criteria and indicators are interrelated. Non-metric multidimensional scaling is usually employed when standards or reference points

Evaluation method	Assumptions/Limitations
Analytic hierarchy process	- Highly dependent on the experience and knowledge of the evaluators/decision-makers
Weighted sum model	<ul> <li>All criteria are measured on cardinal scales and expressed in comparable units; weights are assigned per criterion;</li> <li>Considers independence of criteria;</li> <li>Assumes transitivity of preferences (additive utility);</li> <li>Applicable to single-dimensional problem (all units are similar)</li> </ul>
Ordination technique (e.g., multidimensional scaling)	<ul><li>Useful when dealing with large amount of variables;</li><li>The goal is to quantify the relationships among interdependent variables</li></ul>
Concordance analysis	<ul> <li>Not based on utility theory;</li> <li>Method can be used in cardinal data (impact matrix and weight vector);</li> <li>Can be applied even if evaluation matrix is ordinal as long as the weight vector is cardinal</li> </ul>
Regime method	- Useful when evaluation matrix and weight vector are both ordinals
Mixed data evaluation (Evamix)	- Deals with mixed qualitative-quantitative evaluation scores

 Table 1
 Assumptions and limitations in using the multi-criteria evaluation methods

for criteria or indicators are available. These reference points are considered 'ideal points' such that choice possibilities which highly deviate from the ideal points are located in space farther from the ideal points. In contrast, concordance analysis, though it does not assume independence of criteria and indicators, also requires both criteria scores and weights to be measured on a cardinal scale. Concordance analysis also provides that even if the evaluation matrix is ordinal, the concordance set can still be determined as long as the weight vector is cardinal. However, when both criteria scores and weights are ordinal, then the regime method is appropriate. The appraisal scores in concordance analysis and the regime method are derived through pairwise comparisons of choice possibilities.

This non-exhaustive literature review shows that among the aggregation approaches discussed here, the analytic hierarchy process and ordination technique had the highest number of applications in fisheries while none was recorded for concordance analysis, the regime method or Evamix.

Several studies not only make use of one method but a hybrid of methods. For example, Moriki and Karydis (1994) applied concordance analysis and the regime method to assess eutrophication levels in the coastal ecosystem. They found that both methods produced similar results in ranking alternative sampling locations. Moriki et al. (1996) realized that the regime method and multidimensional scaling [together with cluster analysis and principal component analysis] are applicable to a set of heterogenous data [measured on both ordinal and cardinal scales]. Ridgley and Rijsberman (1994) used the weighted sum model, concordance analysis and the analytic hierarchy process to elucidate the implications of using a set of impact assessments and preference evaluations. In order to analyze a set of criteria with varied units of measurements, Bodini and Giavelli (1992) applied weighted concordance analysis and Evamix. Scores for the concordance analysis were derived quantitatively then converted into qualitative data for the Evamix analysis. Other recent studies which combined multicriteria evaluation models with other decision support tools include the following:

- (a) Zhao and Yang (2009) combined a fuzzy assessment method and hierarchical analysis process to establish the integrative fuzzy hierarchical assessment model and applied it to the case study of Yong River in Ningbo City.
- (b) Hossain et al. (2009), Buitrago et al. (2005), and Kitsiou et al. (2002) used multi-criteria evaluation in combination with geographic information systems (GIS) to identify suitable sites for carp farming, oyster aquaculture, and to rank coastal areas for development, respectively.
- (c) Brody et al. (2006) applied multiple-criteria decision-making and spatial decision support systems in developing an evaluation approach that will identify locations for oil and gas activities in the coastal margin of Texas. The approach was also supported by the use of GIS.

- (d) Adrianto et al. (2005) used a multi-criteria approach to assess the local sustainability of fisheries system in Yoron Island, Kagoshima. They developed a mixed-method approach which used a modified weighting scale of Saaty, a sustainability index of criteria (SIC), and a cognitive mapping technique.
- (e) Leung et al. (2001) used a multi-objective programing model (decision maps, compromise programing and an I-O model) to examine tradeoffs between various conflicting objectives (regional employment, income and economic rent) of the North Norwegian cod fisheries in the Barents sea.

# 4 Application of Multi-Criteria Evaluation Methods to Tropical Fisheries Management

In order to examine the performance of the methods discussed in this paper and to determine their level of agreement in a tropical fisheries situation, a case study is presented. The aim of the case study is to determine the impacts of fisheries management strategies in a Philippine Bay.

# 4.1 Case Study: Evaluation of Fisheries Management Strategies in San Miguel Bay, Philippines

#### 4.1.1 Study Area

San Miguel Bay is considered as one of the most productive fishing grounds in the Philippines (Mines, 1982). However, similar to any rich fishing area, it has been subjected to increased in-migration of people, intensive fishing, use of destructive fishing methods such as dynamite and cyanide, habitat destruction, and intense conflicts among various users, pollution, etc. The bay is situated at the southeastern part of Luzon between longitudes 122°59'E and 123°20'E and latitudes 13°40'N and 14°09'N (Cinco et al., 1995; Sia III and Luna, 1992). It is bordered by the provinces of Camarines Norte on the northwest and Camarines Sur on the south and east borders with a total area of about 1 115 km<sup>2</sup> and a coastline of 188 km (Garces et al., 1995). Seven coastal municipalities surround the bay; namely, Cabusao, Calabanga, Sipocot, Siruma, Tinambac, Basud and Mercedes.

The bay harbors a variety of finfishes and invertebrates. According to Pauly (1982), because of the bay's estuarine environment, 91 euryhaline marine species which tolerate freshwater and/or brackish water were found. He recorded that from 1868 to 1981, 188 species of fish belonging to 71 families were found in the bay. Such diverse fauna became the target of overexploitation as early as the 1970s, as noted by Simpson (1978) cited in Mines (1982). The fishers of the bay use a wide range of traditional fishing gears and methods and these are documented in Pauly et al. (1982) with three additional methods (i.e., the use of explosives in fishing, use of cyanide and other poisons in fishing, and the use of Danish seines (buli-buli)) reported by Silvestre and Cinco (1992). The bay is considered as a traditional ground for trawlers because of its wide area of soft bottoms (Silvestre and Cinco, 1992). During the 1970s, conflicts transpired between gillnetters and commercial trawlers, leading to the banning of commercial trawling within the bay's municipal waters (Mines, 1982).

Multi-disciplinary research in the 1980s already recommended management interventions. However, nothing was done then because of the government's thrust to modernize the fishing industry in the country. In the early part of the 1990s, the bay was revisited through the implementation of the fisheries sector program (FSP) of the Department of Agriculture. The program not only gathered technical information on the status of the bay's resources and users but also implemented resource management interventions that would lead to the bay's recovery. Aside from the programs and projects of the national government, there were also management interventions initiated and implemented by the local government.

#### 4.1.2 Data Collection

The information required were collected through survey interviews and review of secondary sources including records from government offices and outputs of major fisheries research programs and projects in the Philippines. The literature guided the identification and definition of criteria and indicators.

#### 4.1.3 Definition of Fisheries Management Strategies

A fisheries management strategy is composed of several management interventions. In this case study, a coastal municipality (composed of management interventions) represents a particular management strategy, as follows:

- Basud- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, mangrove reforestation, artificial reef deployment, marine fishery reserves/fish sanctuaries
- Cabusao- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, mangrove reforestation, artificial reef deployment
- Calabanga- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, mangrove reforestation, artificial reef deployment, marine fishery reserves/fish sanctuaries
- Mercedes- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, mangrove reforestation, artificial reef deployment, marine fishery reserves/fish sanctuaries
- Sipocot- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, artificial reef deployment
- Siruma- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, mangrove reforestation, artificial reef deployment, marine fishery reserves/fish sanctuaries
- Tinambac- ban on commercial fishing, gear and vessel restrictions, modification of licensing system, implementation of livelihood, mangrove reforestation, artificial reef deployment

#### 4.1.4 Weights and Scores of Criteria and Indicators

There are 24 indicators which belong to five multiple criteria; namely, acceptability, biotic diversity, economic performance, enforceability and equity Table 2. The indicators can be categorized as *cardinal* or *ordinal* indicators. The cardinal weights of the criteria and indicators were elicited from the different sectors of coastal resource users in San Miguel Bay, specifically the institutionalized group known as fisheries and aquatic resource management councils (FARMCs) using the analytic hierarchy process. FARMCs are created in all coastal cities/municipalities of the country pursuant to the provisions of the Republic Act No. 8550, also known as the Philippine Fisheries Code of 1998.

The measures of the indicators are called indicator scores. They were derived from existing data (e.g. databank of the Philippine Fisheries Sector Program) and direct assessment by resource users. The indicator scores were determined at two time periods - before and after implementation of fisheries management strategies. The scores of the 12 ordinal indicators were determined from the members of the Barangay fisheries and aquatic resource management councils (BFARMCs) in each coastal municipality of San Miguel Bay. The 211 BFARMC representatives are knowledgeable, experienced and have lived in the coastal site for years, thus making them a good choice. About 98% of them attended school; 49% had an average of 6 years experience in coastal resource management; and 60% have lived in the municipality since birth. The scores of the 12 cardinal indicators were determined from secondary sources. Most secondary information available for San Miguel Bay is aggregated; thus, one of the challenges in measuring these indicators is the process of disaggregating them so that each municipality would have its own indicator information. All indicators are benefit indicators except for the number of commercial fishing boats and banned fishing gears; employment structure of small-scale fishers; and profit distribution among different fishing gears, which are considered cost indicators.

The 12 ordinal indicators specified the maximum and minimum values at +10 and -10, respectively. And as for cardinal indicators, the maximum and minimum values vary per indicator and would depend on the municipalities. The final output is an impact evaluation matrix showing the degree or level of change. The linear scale transformation was used to normalize the data. It considers maximum and minimum values and these values can be determined in two ways: (a) specify the possible maximum and minimum values of the indicators (values are constant); and (b) use the highest and lowest values from among the

Table 2	Measurement	of	criteria	and	indicators
---------	-------------	----	----------	-----	------------

Criteria	Indi	cators	Unit of measurement	Type of measurement
Acceptability	(a)	Resource users' participation in the	0–10	Ordinal
	(b)	Level of awareness of resource users in fisheries resource management	0–10	Ordinal
	(c)	Number of fishers who belong to an organization	Total number	Cardinal
	(d)	Change in the level of intra-sectoral conflicts	0–10	Ordinal
	(e)	Change in the level of inter-sectoral conflicts	0–10	Ordinal
Biotic diversity	(f)	Abundance of reef fishes	Total frequency of reef fish	Cardinal
	(g)	Abundance of commercial fish catch fish catch	Weight in tons of commercial	Cardinal
	(h)	Species richness of reef fish	Species richness (H')	Cardinal
	(i)	Extent of mangrove areas	Hectares	Cardinal
	(j)	Status of coral reef resources	Percentage live coral cover	Cardinal
Economic performance	(k)	Number of commercial fishing boats and banned fishing gears	Total number	Cardinal
	(1)	Fisherfolk gross revenue from fishing	Total value of fish catch (PhP)	Cardinal
	(m)	Assessment of fisherfolk gross revenue from fishing	0–10	Ordinal
	(n)	Employment structure of small-scale fisheries	Ratio of full-time to part-time fishers	Cardinal
Enforceability	(0)	Presence of comprehensible laws and regulations related to management	0–10	Ordinal
	(p)	Frequency of information dissemination about the management	0–10	Ordinal
	(q)	Perception of suitability of enforcement techniques	0–10	Ordinal
	(r)	Performance assessment of fisheries law enforcers	0–10	Ordinal
	(s)	Financial support for fisheries law	Budget estimates (PhP) enforcement	Cardinal
	(t)	Assessment of the allocated financial support for enforcement	0–10	Ordinal
Equity	(u)	Profit distribution among different fishing gears	Percentage contribution of major fishing gears to the total gross profits	Cardinal
	(v)	Amount of financial support for additional livelihood	Budget (PhP)	Cardinal
	(w)	Assessment of the success of additional livelihood implemented	0–10	Ordinal
	(x)	Inclusion of women in the management process	0–10	Ordinal

seven coastal municipalities being compared (values are variable). The normalized values of the indicators ranged between 0 and 1.0. A particular management strategy (represented by a municipality) is said to perform better with respect to a given indicator when its normalized value is highest or equal to 1.0. Table 3 shows the normalized impact evaluation matrix.

Indicators		Basud	Cabusao	Calabanga	Mercedes	Sipocot	Siruma	Tinambac	
(a)	Resource users' participation in the management process	0.73	0.7	0.75	0.74	0.69	0.85	0.74	
(b)	Level of awareness of resource users in resource management	0.85	0.69	0.76	0.78	0.8	0.79	0.74	
(c)	Number of fishers who belong to an organization	0.01	0.01	0	0.01	0.01	1	0.04	
(d)	Change in the level of intra-sectoral conflicts	0.51	0.49	0.54	0.77	0.41	0.39	0.58	
(e)	Change in the level of inter-sectoral conflicts	0.38	0.3	0.39	0.58	0.3	0.22	0.35	
(f)	Abundance of reef fish				1		0		
(g) (h)	Abundance of commercial fish catch Species richness of reef fish	0.58	0.23	0.68	0.75 1	0.38	0.42 0	0.76	
(i)	Extent of mangrove areas	0.07	0.27	0.72	0.07		0.03	0.33	
(j)	Status of coral reef resources				0.42		0.35		
(k)	Number of commercial fishing boats & banned fishing gears	0.9	0.63	0.18	1	0.98	0	0.6	
(l)	Fisherfolk gross revenue from fishing	0.7	0.22	0.36	0.95	0.27	0.38	0.62	
(m)	Assessment of fisherfolk gross revenue from fishing	0.1	0.15	0.25	0.22	0.18	0.17	0.2	
(1)	Employment structure of small-scale fishers		0	1					
(n)	Presence of comprehensible laws and regulations	0.62	0.64	0.75	0.75	0.65	0.78	0.7	
(p)	Frequency of information dissemination about the management	0.72	0.63	0.78	0.79	0.62	0.65	0.63	
(q)	Perception of the suitability of enforcement techniques	0.89	0.9	0.74	0.83	0.95	0.9	0.84	
(r)	Performance assessment of law enforcers	0.53	0.43	0.38	0.56	0.51	0.48	0.52	
(s)	Financial support for enforcement		0.29	0.9	0.78			0.64	
(t)	Assessment of the allocated financial support for enforcement	0.51	0.52	0.6	0.68	0.52	0.57	0.54	
(u)	Profit distribution among different fishing gears	0	0.97	0.88	0.79	1	0.8	0.94	
(v)	Financial support for additional livelihood implemented		0.7	1	0			0.22	
(w)	Assessment of the success of additional livelihood implemented	0.7	0.66	0.68	0.63	0.67	0.71	0.73	
(x)	Inclusion of women in the management process	0.76	0.72	0.67	0.72	0.76	0.76	0.7	

**Table 3** Normalized impact evaluation matrix

#### 4.1.5 Results

The impacts of fisheries management strategies (represented by the coastal municipalities) were evaluated using concordance analysis, the regime method, Evamix and non-metric multidimensional scaling (MDS). A final outranking of the municipalities was created. Although there may be variability in the ranking for some municipalities, Table 4 shows that the municipality of Mercedes has the optimal fisheries management strategy in all three methods. This means that the management strategy in Mercedes performs the best on the greatest number of indicators. The other municipalities that are not outranked are Siruma and Calabanga. The results show that all three municipalities have marine reserves/fish sanctuaries as a form of management interventions. These are not found in the municipalities of Cabusao, Sipocot and **Table 4** Results ofconcordance analysis, regimemethod and Evamix

Fisheries			Evamix	
management strategies/Coastal municipalities	Concordance analysis	Regime	Subtractive summation technique	Subtracted shifted interval technique
Basud	-2.28	+++	-0.43	-0.04
Cabusao	-0.94	++	-0.86	-1.18
Calabanga	2.75	++	0.29	0.25
Mercedes	4.90	++++++	0.57	0.41
Sipocot	-3.43	++++	-0.14	0.10
Siruma	0.34	+++	0.43	0.26
Tinambac	-1.34	+	0.14	0.19



Fig. 1 Configuration derived in two dimensions

Tinambac. The results of concordance analysis, the regime method and Evamix are comparable, at least for the top three municipalities.

When the data were further examined using nonmetric MDS through SPSS version 11, the results grouped the municipalities into a two-dimensional space. Figure 1 shows that Cabusao, Tinambac and Sipocot are closer to each other in the configuration. In order to determine which indicators influence such association, linear multiple regression was done. The indicator as a dependent variable was regressed over the independent variables which are the coordinates of the configuration. The indicators which provided significant regression results (p < 0.05) include extent of mangrove areas; number of commercial fishing boats & banned fishing gears; employment structure of small-scale fishers; and inclusion of women in the management process. The stress in the configuration is 2.8%, a somewhat good goodness of fit, and thus there is no need for higher dimensionality.

#### 4.1.6 Discussion

The case study applied more than one technique in order to evaluate the impacts of fisheries management strategies. The cardinal weight vector was derived using the analytic hierarchy process while the aggregation procedure applied concordance analysis, the regime method and Evamix. Important issues have to be considered in choosing the appropriate combination of techniques in fisheries impact evaluation. Only a few studies have explored the use of the AHP in fisheries management, maybe because in many impact evaluation studies, coastal resource users' assessment is seldom considered in a formal evaluation process. Many researchers are quite apprehensive about integrating subjective judgments into the process of impact assessment because judgment varies among individuals and with time. This may be one major limitation in terms of integrating resource users into the evaluation process. However, due to the uncertainty and incompleteness of technical information, coastal resource users' experience and local knowledge are critical in providing balance to the entire evaluation process.

Most fisheries criteria (e.g., equity, economic efficiency, ecological sustainability) are broadly defined, and thus would require measurement indicators. Measurement of a criterion is rarely possible with only a single indicator because other indicators may also contribute to the measure of a criterion. There will be instances when the number of indicators is not equal for all criteria (e.g., five indicators for biological diversity

versus four indicators for economic efficiency). If this is the case, Voogd (1983) suggested aggregating indicators per criterion. The simplest form of aggregation is to take the mean of the normalized indicators. Thus, the mean is now considered as the normalized criterion score. Another important consideration is the possibility of finding heterogenous (measured on varying scales) indicators in each criterion. For example, indicator scores for the criterion economic efficiency criterion may be a combination of ordinal, ratio or interval data. In the Evamix method, the criteria or indicators measured quantitatively are separated from those measured qualitatively. Thus, concern as to the validity of combining mixed information is resolved in this method. Its most important feature is a separate calculation of standard dominance scores for ordinal and cardinal criteria/indicators. When the impact evaluation matrix is composed of several criteria/indicators measured on varying scales, Evamix may be a useful aggregating approach. Since Evamix does not specify the minimum number of criteria/indicators to use, then it is possible to use all of them. Voogd (1983) suggested, however, that in any impact evaluation the number of criteria/indicators should be delimited to approximately seven or eight.

# 4.2 Extent of Application of Multi-Criteria Methods to Tropical Fisheries Management

If possible, the evaluation of the impacts of fisheries management should relate to the goals or objectives of fisheries. These objectives are most likely stated in a management plan, inferred from general policy statements or sometimes proposed by the authors (Chesson et al., 1999). The evaluation process, however, becomes very difficult if management objectives are not clearly specified (McAllister et al., 1999). The gaps, difficulties and challenges in developing and implementing fisheries management systems, recognized in an ICES Symposium held on November 16–19, 1998, set out important challenges in the development of a multi-criteria evaluation method for fisheries. The salient points raised during the Symposium, found in Stokes et al. (1999), are summarized as follows:

- a) Evaluation and management of fisheries systems require sound decision-making despite uncertainty. Fisheries management systems must develop techniques to account for these uncertainties;
- b) The slow pace of fisheries management to recognize the need to implement formal and rigorous decision-making. The existing case studies and techniques and approaches from the fields of operation research and management science may be useful;
- c) The need for collaborative efforts among stakeholders (e.g., management agencies, scientists, industry, etc.) to articulate objectives for fisheries management to be consistent with international fishery conventions and standards;
- d) There is a change in the governance for fisheries, from a single discipline to a multi-discipline approach involving the socio-economic context of the fishery; and
- e) The relevance of formal evaluation and management procedures and system performance in providing information upon which credible management decisions can be based.

The outcome of the conference articulated the need for and lack of an evaluation instrument for fisheries management systems. The direction is no longer towards a single disciplinary approach but is multi-disciplinary with respect to what is to be evaluated and who will participate in the evaluation.

While multi-criteria methods have been increasing in application to natural resource management decision problems, their application to fisheries management appears to have been slow. The best-known examples are found in the reviews of Romero and Rehman (1987) and Mardle and Pascoe (1999). Romero and Rehman (1987) reviewed 13 case studies on how multi-criteria decision models [i.e., multi-objective programing (MOP), lexicographic goal programing (LGP), weighted goal programing (WGP) and multiattribute utility function (MUF)] were applied in fisheries. They found that more than half focused on the establishment of the optimum structure of the fishing fleet in particular geographical areas. Mardle and Pascoe (1999) found 30 case studies relating to management schemes for a particular fishery; fleet structure analysis, requirements and composition; policy and development plans; catch quota policies; resource

Model	Decision problem	Authors
Analytic hierarchy process (AHP)	Evaluate fisheries management options	Fernandes et al. (1999); Nielsen and Mathiesen (2006); Pascoe et al. (2009); Soma (2003)
	Measure importance of management objectives and performance indicators	Himes (2007); Mardle et al. (2004); Utne (2008); Whitmarsh and Palmieri (2009)
	Site selection	Ramos et al. (2006)
AHP + CADS_TOOL (cage aquaculture decision support tool)	Evaluate cage aquaculture site	Halide et al. (2009)
Multiple objective programing (MOP), compromise programing (CP), weighted goal programing (WGP)	Planning for regional aquaculture development	El-Gayar and Leung (2001)
Fuzzy logic model of expert knowledge	Evaluate ecosystem performance of the South African sardine Sardinops sagax fishery	Paterson et al. (2007)
MCA, modified AHP, sustainability index of criteria (SIC), cognitive mapping technique	Assess local sustainability of fisheries system in Yoron Island, Kagoshima	Adrianto et al. (2005)
Multi-objective programing model (decision maps, compromise programing, I-O model)	Tradeoffs between various conflicting objectives of the North Norwegian cod fisheries in the Barents sea	Leung et al. (2001)
Multi-criteria evaluation (MCE) + geographical information systems (GIS)	Identify suitable sites for aquaculture species carp farming in Chittagong, Bangladesh	Hossain et al. (2009); Buitrago et al. (2005)
Weighted sum model	Fisheries management evaluation	Chesson et al. (1999)
Ordination technique (marine protected area evaluation model (MPAEM) and rapid appraisal technique for fisheries (Rapfish))	Evaluate the status of fisheries globally	Alder et al. (2002); Alder et al. (2000); Baeta et al. (2005); Murillas et al. (2008); Pitcher and Preikshot (2001); Pitcher et al. (2009); Pitcher (1999); Preikshot et al. (1998); Tesfamichael and Pitcher (2006)

**Table 5** Multi-criteria evaluation method applications in fisheries

allocation; fishing effort input; optimal harvesting; commercial sampling schemes; fishing sites; and limiting entry. These case studies applied linear goal programing, generating methods, non-linear programing, multi-attribute utility theory, the analytic hierarchy process and multi-level Programing. Based on the present review of papers drawn from peer-reviewed scientific journals, there are 26 papers in fisheries that have applied multi-criteria evaluation methods during the past 10 years and these are summarized in Table 5.

### 5 Conclusion

This paper reveals the relatively few applications of multi-criteria decision models to tropical fisheries management. While they show potential because of their ability to: (a) provide a balance and integrate the various components of fisheries encompassing ecological, biological, social, economic and institutional objectives; (b) incorporate judgments of the various stakeholders in fisheries; (c) handle mixed information; and (d) allow interactions between the objective and subjective measures of the criteria and indicators, they are not extensively used. The main reason for such non-extensive application is because fisheries management evaluation has always been treated according to discipline, and not in an integrated manner. To measure the outcomes of management in a holistic or integrated manner is often problematic and therefore, only fragments of change are usually dealt with (Hanson, 2003) because there are restrictions and limitations in the measurement of change. Contributing to the limitations is the sense that each discipline (e.g., ecological/biological, social, political)

Factors	Limitations
Data management	<ul> <li>too laborious unless a computer program is developed to make computation easier;</li> </ul>
	<ul> <li>although not seen as a crucial factor, complete data for the indicators are ideal to increase the reliability of results;</li> </ul>
	<ul> <li>supplied data may not contain all the information that the stakeholders/decision-makers conveyed</li> </ul>
Acceptability of the approach	<ul> <li>new in the field of fisheries management, therefore the approach has to evolve into a more simplified one (i.e., the basic rule is that it should be easily understood)</li> </ul>
Capability and willingness of stakeholders/resource users to apply the methods	<ul> <li>resource users with technical and analytical knowledge may be able to apply the multi-criteria approach but they would require extensive training;</li> <li>sometimes resource users/stakeholders may not be willing to participate, especially when the process requires considerable time</li> </ul>
Technical and financial constraints	<ul> <li>robustness and reliability of the method;</li> <li>multi-criteria evaluation is quite expensive because it involves participation of different stakeholders and extensive data collection; thus, this should be part of a major coastal project or program</li> </ul>

Table 6 Factors limiting the application of multi-criteria evaluation models in tropical fisheries management

has its own assumptions that need to be satisfied. For example, Hruby (1999) indicated that statistical properties of decision-making models are different from those of ecological models, i.e., statistical approaches based on analysis of variance and normality of data are not appropriate when mixed qualitative and quantitative data or subjective judgments are incorporated.

Often, political leaders or decision-makers are faced with the problem of what to do with the types of information and how to integrate them so they can become usable for policy-making purposes. The issue of integration, though essential, is quite problematic in fisheries management because of alleged insufficiency in analytical methods. This may be resolved by further examining existing multi-criteria models in the field of operational research that may have potential use in evaluating tropical fisheries management. However, the limits of their application should also be recognized (see Table 6). In general, the choice of method would depend on whether it is appropriate for a particular decision-making problem or it is able to handle the situation correctly considering contextual, technical and political concerns. This is highly important in developing countries wherein cost associated with impact evaluation is a major constraint.

#### References

- Abu-Taleb M.F. (2000) Application of multicriteria analysis to the design of wastewater treatment in a nationally protected area, Environ. Eng. Policy, 2, 37–46.
- Adriaenssens V., Baets B.D., Goethals P.L.M., Pauw N.D. (2004) Fuzzy rule-based models for decision support in ecosystem management, Sci. Total Environ. 319, 1–12.
- Adrianto L., Matsuda Y., Sakuma Y. (2005) Assessing local sustainability of fisheries system: a multi-criteria participatory approach with the case of Yoron Island, Kagoshima prefecture, Jpn Mar. Policy 29, 9–23.
- Agardy T. (2000) Effects of fisheries on marine ecosystems: a conservationist's perspective, ICES J. Marine Sci. 57, 761–765.
- Alder J., Pitcher T.J., Preikshot D., Kaschner K., Ferriss B. (2000) How good is good? A rapid appraisal technique for evaluation of the sustainability status of fisheries of the North Atlantic, in: Pauly D., Pitcher T.J. (Eds.), Methods for evaluating the impacts of fisheries on North Atlantic ecosystems. Fisheries Centre Research Reports, University of British Columbia, Canada, Vol. 8, No. 2, pp. 136–182.
- Alder J., Zeller D., Pitcher T., Sumaila R. (2002) A method for evaluating marine protected area management, Coast. Manage. 30, 121–131.
- Allison E.H., Ellis F. (2001) The livelihoods approach and management of small-scale fisheries, Mar. Policy 25, 377–388.
- Anderson L.G. (1989) Enforcement issues in selecting fisheries management policy, Mar. Res. Econ. 6, 261–277.
- Ascough II J.C., Maier H.R., Ravalico J.K., Strudley M.W. (2008) Future research challenges for incorporation of
uncertainty in environmental and ecological decisionmaking, Ecol. Model. 219, 383-399.

- Baeta F., Pinheiro A., Corte-Real M., Lino Costa J., Raposo de Almeida P., Cabral H., Costa M.J. (2005) Are the fisheries in the Tagus estuary sustainable? Fish. Res. 76, 243–251.
- Banai R. (2005) Land Resource Sustainability for Urban Development: Spatial Decision Support System Prototype, Environ. Manage. 36, 282–296.
- Becker H.A. (2001) Social impact assessment, Eur. J. Oper. Res. 128, 311–321.
- Berkes F., Mathias J., Kislalioglu M., Fast H. (2001) The Canadian Arctic and the Oceans Act: the development of participatory environmental research and management, Ocean Coast. Manage. 44, 451–469.
- Bingham G., Bishop R., Bromley M.B.D., Costanza E.C.W.C.R., Hale T., Hayden G., Kellert S., Norgaard R., Norton B., Payne J., Russell C., Suter G. (1995) Issues in ecosystem valuation: improving information for decisionmaking, Ecol. Econ. 14, 73–90.
- Bodini A., Giavelli G. (1992) Multicriteria analysis as a tool to investigate compatibility between conservation and development on Salina Island, Aeolian Archipelago, Italy, Environ. Manage. 16, 633–652.
- Bojórquez-Tapia L.A., Sánchez-Colon S., Martinez A.F. (2005) Building consensus in environmental impact assessment through multicriteria modeling and sensitivity analysis, Environ. Manage. 36, 469–481.
- Bonzon A. (2000) Development of economic and social indicators for the management of Mediterranean fisheries, Mar. Freshwater Res. 51, 493–500.
- Bowen R.E., Riley C. (2003) Socio-economic indicators and integrated coastal management, Ocean Coast. Manage. 46, 299–312.
- Brody S.D., Grover H., Bernhardt S., Tang Z., Whitaker B., Spence C. (2006) Identifying Potential Conflict Associated with Oil and Gas Exploration in Texas State Coastal Waters: A Multicriteria Spatial Analysis, Environ. Manage. 38, 597–617.
- Bryan B.A., Crossman N.D. (2008) Systematic regional planning for multiple objective natural resource management, J. Environ. Manage. 88, 1175–1189.
- Buitrago J., Rada M., Hernandez H., Buitrago E. (2005) A Single-Use Site Selection Technique, Using GIS, for Aquaculture Planning: Choosing Locations for Mangrove Oyster Raft Culture in Margarita Island, Venezuela, Environ. Manage. 35, 544–556.
- Bundy A., Pauly D. (2001) Selective harvesting by small-scale fisheries: ecosystem analysis of San Miguel Bay, Philippines, Fisheries Res. 53, 263–281.
- Caddy J.F., Csirke J., Garcia S.M., Grainger R.J.R. (1998) How Pervasive is "Fishing Down Marine Food Webs"? Science 282, 1383a.
- Cairns J.J., McCormick P.V., Niederlehner B.R. (1993) A proposed framework for developing indicators of ecosystem health, Hydrobiologia 1–44.
- Castilla J.C., Fernandez M. (1998) Small-scale fisheries in Chile: On co-management and sustainable use of benthic invertebrates, Ecol. Appl. 8, S124–132.
- Charles A.T. (1989) Bio-socio-economic fishery models: labor dynamics and multiobjective management, Can. J. Fish. Aquat. Sci. 46, 1313–1322.

- Chesson J., Clayton H., Whitworth B. (1999) Evaluation of fisheries-management systems with respect to sustainable development, ICES J. Mar. Sci. 56, 980–984.
- Chong K.C. (2000) Using sustainability indicators to manage fisheries: experiences on the Bay of Bengal, Mar. Freshwater Res. 51, 523–527.
- Chung E.-S., Lee K.S. (2009) Identification of Spatial Ranking of Hydrological Vulnerability Using Multi-Criteria Decision Making Techniques: Case Study of Korea, Water Resour. Manage., DOI 10.1007/s11269-008-9387-9.
- Cinco E.A., Mendoza D.J., Ricafrente-Remoto M.C. (1995) Results of bathymetric survey in San Miguel Bay, in: Silvestre G., Luna C., Padilla J. (Eds.), Multidisciplinary Assessment of the Fisheries in San Miguel Bay, Philippines, ICLARM Technical Report 47, CD-ROM, Manila.
- Clarke S., Leung A.W.-Y., Mak Y.M., Kennish R., Haggan N. (2002) Consultation with local fishers on the Hong Kong artificial reefs initiative, ICES J. Mar. Sci. 59, S171–S177.
- Cochrane K.L. (2000) Reconciling sustainability, economic efficiency and equity in fisheries: the one that got away? Fish and Fisheries 1, 3–21.
- Crawford B.R., Siahainenia A., Rotinsulu C., Sukamra A. (2004) Compliance and Enforcement of Community-Based Coastal Resource Management Regulations in North Sulawesi, Indonesia, Coast. Manage. 32, 39–50.
- Dahl A.L. (2000) Using indicators to measure sustainability: recent methodological and conceptual developments, Mar. Freshwater Res. 51, 427–433.
- Dalton T.M. (2004) An approach for integrating economic impact analysis into the evaluation of potential marine protected area sites, J. Environ. Manage. 70, 333–349.
- Dey P.K., Ramcharan, E.K. (2008) Analytic hierarchy process helps select site for limestone quarry expansion in Barbados, J. Environ. Manage. 88, 1384–1395.
- DiNardo G., Levy D., Golden B. (1989) Using decision analysis to manage Maryland's river herring fishery: An application of AHP, J. Environ. Manage. 29, 192–213.
- Dinmore T.A., Duplisea D.E., Rackham B.D., Maxwell D.L., Jennings S. (2003) Impact of a large-scale area closure on patterns of fishing disturbance and the consequences for benthic communities, ICES J. Mar. Sci. 60, 371–380.
- Dixon J.A., Scurra L.F., Carpenter R.A., Sherman P.B. (1994) Economic Analysis of Environmental Impacts (New Edition), Earthscan Publication Ltd, London.
- Duke J.M., Aull-Hyde R. (2002) Identifying public preferences for land preservation using the analytic hierarchy process, Ecol. Econ. 42, 131–145.
- Eagle J., Thompson Jr. B.H. (2003) Answering Lord Perry's question: dissecting regulatory overfishing, Ocean Coast. Manage. 46, 649–679.
- Ehler C.N. (2003) Indicators to measure governance performance in integrated coastal management, Ocean Coast. Manage. 46, 335–345.
- El-Gayar O., Leung P. (2001) A multiple criteria decision making framework for regional aquaculture development, Eur. J. Oper. Res. 133, 462–482.
- FAO Fisheries and Aquaculture Department (2009) The State of World Fisheries and Aquaculture 2008. Food and Agriculture Organization of the United Nations, Rome, 178 p.

- Fernandes L., Ridgley M.A., Hof T.v.t. (1999) Multiple criteria analysis integrates economic, ecological and social objectives for coral reef managers, Coral Reefs 18, 1–10.
- Ferrer E.M., Cruz L.P.d., Domingo M.A. (1996) Seeds of Hope: A Collection of Case Studies on Community-based Coastal Resources Management in the Philippines, College of Social Work and Community Development, University of the Philippines, Diliman, Quezon City, Philippines and NGO Technical Working Group for Fisheries Reform and Advocacy (NGO TWG), Quezon City.
- Froese R. (2004) Keep it simple: three indicators to deal with overfishing, Fish and Fisheries 5, 86–91.
- Garces L.R., Vega M.J.M., Ledesma R.G.G., Sia III Q.P., Robles N.A. (1995) San Miguel Bay coastal habitat mapping and assessment using remote sensing techniques in: Silvestre G., Luna C., Padilla J. (Eds.), Multidisciplinary assessment of the fisheries in San Miguel Bay, Philippines (1992–1993), ICLARM Tech. Rep. 47, CD-ROM, Manila.
- Garcia S.M., Staples D.J. (2000) Sustainability reference system and indicators for responsible marine capture fisheries: a review of concepts and elements for a set of guidelines, Mar. Freshwater Res. 51, 385–426.
- Garcia S.M., Staples D.J., Chesson J. (2000) The FAO guidelines for the development and use of indicators for sustainable development of marine capture fisheries and an Australian example of their application, Ocean Coast. Manage. 43, 537–556.
- Gilman E.L. (1997) Community based and multi purpose protected areas: A model to select and manage protected areas with lessons from the Pacific Islands, Coast. Manage. 25, 59–91.
- Gislason H., Sinclair M., Sainsbury K., O'Boyle R. (2000) Symposium overview: incorporating ecosystem objectives within fisheries management, ICES J. Mar. Sci. 57, 468–475.
- Gregory R. (1987) Nonmonetary measures of nonmarket fishery resources, Transactions of the American Fisheries Society 116, 374-380.
- Griffis R., Kimball K. (1996) Ecosystem approaches to coastal and ocean stewardship, Ecol. Appl. 6, 708–712.
- Hajkowicz S., Collins K. (2007) A Review of Multiple Criteria Analysis for Water Resource Planning and Management, Water Resour. Manage. 21, 1553–1566.
- Hajkowicz S., Higgins A. (2008) A comparison of multiple criteria analysis techniques for water resource management, Eur. J. Oper. Res. 184, 255–265.
- Hajkowicz S.A. (2008) Supporting multi-stakeholder environmental decisions, J. Environ. Manage. 88, 607–614.
- Halide H., Stigebrandt A., Rehbein M., McKinnon A.D. (2009) Developing a decision support system for sustainable cage aquaculture, Environ. Modell. Softw. 24, 694–702.
- Hanson A.J. (2003) Measuring progress towards sustainable development, Ocean Coast. Manage. 46, 381–390.
- Hardi P., Barg S., Hodge T., Pinter L. (1997) Measuring sustainable development: Review of current practice, Occasional Paper 17, 119.
- Harte, M.J., Lonergan S.C. (1995) A multidimensional decisionsupport approach to sustainable development planning, Int. J. Sust. Dev. World 2, 86–103.
- Heen K. (1989) Impact analysis of multispecies marine resource management, Mar. Res. Econ. 6, 331–348.

- Herath G. (2004) Incorporating community objectives in improved wetland management: the use of the analytic hierarchy process, J. Environ. Manage. 70, 263–273.
- Himes A. (2007) Performance indicator importance in MPA management using a multi-criteria approach, Coast. Manage. 35, 601–618.
- Hinloopen E., Nijkamp P., Rietveld P. (2004) Integration of ordinal and cardinal information in multi-criteria ranking with imperfect compensation, Eur. J. Oper. Res. 158, 317–338.
- Ho W. (2008) Integrated analytic hierarchy process and its applications – A literature review, Eur. J. Oper. Res. 186, 211–228.
- Hoehn J.P. (1987) Contingent valuation in fisheries management: the design of satisfactory contingent valuation formats, T. Am. Fish. Soc. 116, 412–419.
- Hossain M.S., Chowdhury S.R., Das N.G., Sharifuzzaman S.M., Sultana A. (2009) Integration of GIS and multicriteria decision analysis for urban aquaculture development in Bangladesh, Landscape Urban Plan. 90, 119–133.
- Hruby T. (1999) Assessments of Wetland Functions: What They Are and What They Are Not, Environ. Manage. 23, 75–85.
- Hundloe T.J. (2000) Economic performance indicators for fisheries, Mar. Freshwater Res. 51, 485–491.
- Hwang C.-L., Yoon K. (1981) Multiple Attribute Decision Making: Methods and Application (A State-of-the-Art Survey), Springer-Verlag, New York.
- Imperial M.T. (1999) Analyzing institutional arrangements for ecosystem-based management: Lessons from the Rhode Island salt ponds SAM plan, Coast. Manage. 27, 31–56.
- Jennings S. (2005) Indicators to support an ecosystem approach to fisheries, Fish and Fisheries 6, 212–232.
- Jensen, C.L. (2002) Reduction of the fishing capacity in "common pool"fisheries, Mar. Policy 26, 155–158.
- Johannes E. (1998) The case for data-less marine resource management: examples from tropical nearshore finfisheries, Tree 13, 243–246.
- Kabuta S.H., Laane R.W.P.M. (2003) Ecological performance indicators in the North Sea: development and application, Ocean Coast. Manage. 46, 277–297.
- Karydis M., Coccosis H. (1990) Use of multiple criteria for eutrophication assessment of coastal waters, Environ. Monit. Assess. 14, 89–100.
- Keeney R.L., Wood E.F. (1977) An illustrative example of the use of multiattribute utility theory for water resources planning, Water Resour. Res. 13, 705–712.
- Khorramshahgol R., Moustakis V.S. (1988) Delphic hierarchy process (DHP): A methodology for priority setting derived from the Delphi method and analytical hierarchy process, Eur. J. Oper. Res. 37, 347–354.
- Khorsid M., Morgan G.R. (1990) A modelling framework for fisheries development planning, Ocean Shoreline Manage. 14, 11–33.
- Kitsiou D., Coccossis H., Karydis M. (2002) Multi-dimensional evaluation and ranking of coastal areas using GIS and multiple criteria choice methods, Sci. Total Environ. 284, 1–17.
- Knuth B.A., Nielsen L.A. (1989) Social and institutional performance indicators for wildlife and fishery resource management systems, Soc. Nat. Resour. 2, 329–344.
- Lane D.E., Stephenson R.L. (1995) A framework for integrated analysis in fisheries manage, INFOR 34, 156–180.
- Legendre L., Legendre P. (1983) Numerical Ecology, Elsevier Scientific Publishing Company, Amsterdam.

- Leung P., Heen K., Bardarson H. (2001) Regional economic impacts of fish resources utilization from the Barents Sea: Trade-offs between economic rent, employment and and income, Eur. J. Oper. Res. 133, 432–446.
- Leung P., Muraoka J., Nakamoto S.T., Pooley S. (1998) Evaluating fisheries management options in Hawaii using AHP, Fish. Res. 36, 171–183.
- Linkov I., Satterstrom F.K., Kiker G., Batchelor C., Bridges T., Ferguson E. (2006) From comparative risk assessment to multi-criteria decision analysis and adaptive management: Recent developments and applications, Environ. Int. 32, 1072–1093.
- Linton, D.M., Warner G.F. (2003) Biological indicators in the Caribbean coastal zone and their role in integrated coast. management, Ocean Coast. Manage. 46, 261–276.
- Lowry G.K., White A.T., Christie P. (2009) Scaling Up to Networks of Marine Protected Areas in the Philippines: Biophysical, Legal, Institutional, and Social Considerations, Coast. Manage. 37, 274–290.
- Mahmoud M.R., Garcia L.A. (2000) Comparison of different multicriteria evaluation methods for the Red Bluff diversion dam, Environ. Modell. Softw. 15, 471–478.
- Makowski M., SomlyÓdy L., Watkins D. (1996) Multiple criteria analysis for water quality management in the nitra basin, Water Resour. Bull. 32, 937–951.
- Maliao R.J., Webb E.L., Jensen K.R. (2004) A survey of stock of the donkey's ear abalone, *Haliotis asinina* L. in the Sagay Marine Reserve, Philippines: Evaluating the effectiveness of marine protected area enforcement, Fish. Res. 66, 343–353.
- Mardle S., Pascoe S. (1999) A review of applications of multi-criteria decision-making techniques to fisheries, Mar. Resour. Econ. 14, 41–63.
- Mardle S., Pascoe S., Herrero I. (2004) Management objective importance in fisheries: an evaluation using the Analytic Hierarchy Process, Environ. Manage. 33, 1–11.
- Mardle S., Pascoe S., Tamiz M., Jones D. (1997) Resource allocation in the North Sea fishery: a goal programming approach, CEMARE Res. Pap. 119, 21.
- Martel J.M., Matarazzo B. (2005) Other Outranking Approaches, in: Figueira J., Greco S., Ehrogott M. (Eds.), Multiple Criteria Decision Analysis: State of the Art Surveys, International Series in Operations Research Management Science, Vol. 78, Springer, New York, pp. 197–259.
- Mawdsley J., O'Malley R. (2009) Development of multi-species indicators for the Nevada Wildlife Action Plan, Ecol. Ind. 9, 1030–1036.
- McAllister M.K., Starr P.J., Restrepo V.R., Kirkwood G.P. (1999) Formulating quantitative methods to evaluate fishery management systems: what fishery process should be modeled and what trade-offs should be made, ICES J. Mar. Sci. 56, 900–916.
- McGarigal K., Cushman S., Stafford S. (2000) Multivariate Statistics for Wildlife and Ecology Research, Springer-Verlag, New York, 283 p.
- McManus J.W. (1986) Depth zonation in a demersal fishery in the Samar Sea, Philippines, in: Maclean J.L., Dizon L.B., Hosillos L.V. (Eds.), First Asian Fisheries Forum, Asian Fisheries Society, Manila, Philippines, pp. 483–486.
- McManus J.W., Reyes R.B. Jr., Nañola C.L. Jr. (1997) Effects of some destructive fishing methods on coral cover and potential rates of recovery, Environ. Manage. 21, 69–78.

- Mines A.N. (1982) The assessment of the fisheries: objectives and methodology, in: Pauly D., Mines A.N. (Eds.), Smallscale fisheries of San Miguel Bay, Philippines: biology and stock assessment, Institute of Fisheries Development and Research, College of Fisheries, University of the Philippines in the Visayas, Quezon City, Philippines; International Center for Living Aquatic Resources Management, Manila, Philippines; and the United Nations University, Tokyo, Japan, Manila, pp. 1–4.
- Moriki A., Karydis M. (1994) Application of multicriteria choice-methods in assessing eutrophication, Environ. Monit. Assess. 33, 1–18.
- Moriki A., Coccossis H., Karydis M. (1996) Multicriteria evaluation in coast management, J. Coastal Res. 12, 171–178.
- Munda G. (2005) Multiple Criteria Decision Analysis and Sustainable Development, in: Figueira J., Greco S., Ehrogott M. (Eds.), Multiple Criteria Decision Analysis: State of the Art Surveys. International Series in Operations Research & Management Science, Vol. 78, Springer, New York, pp. 953–986.
- Munda G., Nijkamp P., Rietveld P. (1994) Qualitative multicriteria evaluation for environmental management, Ecol. Econ. 10, 97–112.
- Murillas A., Prellezo R., Garmendia E., Escapa M., Gallastegui C., Ansuategi A. (2008) Multidimensional and intertemporal sustainability assessment: A case study of the Basque trawl fisheries, Fish. Res. 91, 222–238.
- Myers R.A., Quinn II T.J. (2002). Estimating and testing nonadditivity in fishing mortality: Implications for detecting a fisheries collapse, Can. J. Fish. Aquat. Sci. 59, 597–601.
- Neis B. (2000) In the eye of the storm: Research, activism and teaching with the Newfounland fishery crisis, Women's Studies International Forum 23, 287–298.
- Nickerson D.J. (1999) Trade-offs of mangrove area development in the Philippines, Ecol. Econ. 28, 279–298.
- Nielsen J.R., Mathiesen, C. (2006) Stakeholder preferences for Danish fisheries management of sand eel and Norway pout, Fish. Res. 77, 92–101.
- Nijkamp P., Torrieri F. (2000) A decision support system for assessing alternative projects for the design of a new road network (Methodology and application of a case study), Research Memorandum, 2000-2. ftp://zappa.ubvu.vu. nl/20000021.pdf.
- Nijkamp P., Rietveld P., Voogd H. (1990) Multicriteria Evaluation and Physical Planning, North Holland Publ. Co., Amsterdam.
- Nijkamp P., Vindigni G. (1999) Impact Assessment of Qualitative Policy Scenarios: A Comparative Case Study on Land Use in Sicily, Research Memorandum 1999-4, vrije Universiteit, Amsterdam. ftp://zappa.ubvu.vu.nl/19990004. pdf.
- Parr T.W., Sier A.R.J., Battarbee R.W., Mackay A., Burgess J. (2003) Detecting environmental change: science and society perspectives on long-term research and monitoring in the 21st century, Sci. Total Environ. 310, 1–8.
- Pascoe S., Bustamante R., Wilcox C., Gibbs M. (2009) Spatial fisheries management: A framework for multi-objective qualitative assessment, Ocean Coast. Manage. 52, 130–138.
- Pastorok R.A., MacDonald A., Sampson J.R., Wilber P., Yozzo D.J., Titre J.P. (1997) An ecological decision framework for environmental restoration projects, Ecol. Eng. 9, 89–107.

- Pauly D. (1982) The fishes and their ecology, in: Pauly D., Mines A.N. (Eds.), Small-scale Fisheries of San Miguel Bay: Biology and Stock Assessment, ICLARM Technical Report 7, Institute of Fisheries Development and Research, College of Fisheries, University of the Philippines in the Visayas, Quezon City, Philippines; International Center for Living Aquatic Resources Management, Manila, Philippines; and the United Nations University, Tokyo, Japan, pp. 15–33.
- Pauly D., Mines A.N., Navaluna N.A. (1982) Catch and effort in the small-scale fisheries, in: Pauly, D., Mines A.N. (Eds.), Small-scale Fisheries of San Miguel Bay: Biology and Stock Assessment, ICLARM Technical Report 7, Institute of Fisheries Development and Research, College of Fisheries, University of the Philippines in the Visayas, Quezon City, Philippines; International Center for Living Aquatic Resources Management, Manila, Philippines; and the United Nations University, Tokyo, Japan, pp. 56–64.
- Peterson D.L., Silsbee D.G., Schmoldt D.L. (1994) A case study of resource management planning with multiple objectives and projects, Environ. Manage. 18, 729–742.
- Petry F. (1990) Who is afraid of choices? A proposal for multicriteria analysis as a tool for decision-making support in development planning, J. Int. Dev. 2, 209–231.
- Pitcher T.J. (1999) Rapfish, A Rapid Appraisal Technique For Fisheries, And Its Application To The Code Of Conduct For Responsible Fisheries, FAO Fisheries Circular No. FIRM/C: No. 947, 47 p.
- Pitcher T.J., Preikshot, D. (2001) RAPFISH: a rapid appraisal technique to evaluate the sustainability status of Fisheries, Fish. Res. 49, 255–270.
- Pitcher T.J., Kalikoski D., Short K., Varkey D., Pramoda G. (2009) An evaluation of progress in implementing ecosystem-based managementof fisheries in 33 countries, Mar. Policy 33, 223–232.
- Pollnac R.B., Pomeroy R.S., Harkes I.H.T. (2001) Fishery policy and job satisfaction in three Southeast Asian fisheries. Ocean Coast. Manage. 44, 531–544.
- Pomerol J.C., Barba-Romero S. (2000) Multicriterion Decision in Management: Principles and Practice, Kluwer Academic Publishers, Massachusetts.
- Pomeroy R.S., Pollnac R.B., Katon B.M., Predo C.D. (1997) Evaluating factors contributing to the success of communitybased coastal resource management: the Central Visayas Regional Project-1, Philippines, Ocean Coast. Manage. 36, 97–120.
- Prato T. (1999) Risk-based multiattribute decision-making in property and watershed management, Nat. Resour. Model. 12, 307–334.
- Preikshot D., Nsiku E., Pitcher T., Pauly D. (1998) An interdisciplinary evaluation of the status and health of African lake fisheries using a rapid appraisal technique, J. Fish Biol. 53, 381–393.
- Propst, D.B., Gavrilis, D.G. (1987) Role of economic impact assessment procedures in recreational fisheries management, T. Am. Fish. Soc. 116, 450–460.
- Ramos J., Santos M.N., Whitmarsh D., Monteiro C.C. (2006) The usefulnes of the analytic hierarchy process for understanding reef diving choices: A case study, Bull. Mar. Sci. 78, 213–219.

- Reynolds C. (1993) The ecosystems approach to water management. The main features of the ecosystem concept, J. Aquat. Ecosyst. Health 2, 3–8.
- Rice J. (2003) Environmental health indicators, Ocean Coast. Manage. 46, 235–259.
- Ridgley M.A., Rijsberman F.R. (1992) Multicriteria evaluation in a policy analysis of a Rhine Estuary, Water Resour. Bull. 28, 1095–1110.
- Ridgley M.A., Rijsberman F.R. (1994) Multicriterion analysis and the evaluation of restoration policies for a Rhine estuary, Socio-Econ. Plann. Sci. 28, 19–32.
- Romero C., Rehman T. (1987) Natural resources management and the use of multiple criteria decision-making techniques: A review, Eur. Rev. Agric. Econ. 14, 61–89.
- Roy B. (1991) The outranking approach and the foundations of ELECTRE methods, Theory and Decision 31, 49–73.
- Rutman L. (1984) Evaluation Research Methods: A Basic Guide, Sage Publications, Inc., California, USA.
- Saaty T.L. (1980) The Analytic Hierarchy Process, McGraw-Hill, New York.
- Saaty T.L. (2001) Decision-making for Leaders: The Analytic Hierarchy Process for Decisions in a Complex World, RWS Publications, Pittsburg.
- Salas S., Gaertner D. (2004) The behavioural dynamics of fishers: management implications, Fish and Fisheries 5, 153–167.
- Schmoldt D.L., Peterson D.L., Silsbee D.G. (1994) Developing inventory and monitoring programs based on multiple objectives, Ecol. Manage. 18, 707–727.
- Sia III Q.P., Luna C.Z. (1992) General setting- Biophysical environment, in: Silvestre G., Luna C., Montalvo H. (Eds.), The Coastal Environmental Profile of San Miguel Bay, Philippines, ICLARM, Manila, CD ROM.
- Silvestre G., Cinco E.A. (1992) Capture fisheries sector and coastal resource management issues: San Miguel Bay capture fisherie - An overview, in: Silvestre G., Luna C., Montalvo H. (Eds.), The Coastal Environmental Profile of San Miguel Bay, Philippines, ICLARM, Manila, CD ROM.
- Sinclair M., O'Boyle R. N., Burke D. L., Peacock F.G. (1999) Groundfish management in transition within the Scotia– Fundy area of Canada, ICES J. Mar. Sci. 56, 1014–1023.
- Smith P.G.R., Theberge J.B. (1987) Evaluating natural areas using multiple criteria: theory and practice, Environ. Manage. 11, 445–460.
- Soma K. (2003) How to involve stakeholders in fisheries management – a country case study in Trinidad and Tobago, Mar. Policy 27, 47–58.
- Srdjevic B. (2007) Linking analytic hierarchy process and social choice methods to support group decision-making in water management, Decision Support Systems 42, 2261–2273.
- Stalans L.J. (1995) Multidimensional Scaling, in: Grimm L.G., Yarnold P.R. (Eds.), Reading and Understanding Multivariate Statistics, American Psychological Association, Washington, DC, pp. 137–168.
- Staples D.J. (1997) Indicators of sustainable fisheries development, in: Hancock D.A., Smith D.C., Grant A., Beumer J.P. (Eds.), 2nd World Fisheries Congress, 28 July–2 August 1997, CSIRO, Brisbane, Australia, pp. 719–725.
- Stewart T.J., Scott L. (1995) A scenario-based framework for multicriteria decision analysis in water resources planning, Water Resour. Bull. 31, 2835–2843.

- Stokes T.K., Butterworth D.S., Stephenson R.L. (1999) Confronting uncertainty in the evaluation and implementation of fisheries-management systems, ICES J. Mar. Sci. 56, 795–796.
- Sumaila U.R., Guénette S., Alder J., Chuenpagdee R. (2000) Addressing ecosystem effects of fishing using marine protected areas, ICES J. Mar. Sci. 57, 752–760.
- Sutinen J.G. (1999) What works well and why: evidence from fishery-management experiences in OECD countries, ICES J. Mar. Sci. 57, 752–760.
- Sylvia G., Cai D. (1995) Generating policy information for fisheries management: A comparison of alternative approaches, Mar. Resour. Econ. 10, 77–91.
- Tam J., Palma W., Riofrio M., Aracena O., Lépez M.I. (1996) Decision analysis applied to the fishery of the sea snail *Concholepas concholepas* from the Central Northern Coast of Chile, NAGA, The ICLARM Quarterly, pp. 45–98.
- Tesfamichael D., Pitcher T.J. (2006) Multidisciplinary evaluation of the sustainability of Red Sea fisheries using Rapfish, Fish. Res. 78, 227–235.
- Ticheler H.J., Kolding J., Chanda B. (1998) Participation of local fishermen in scientific fisheries data collection: a case study from the Bangweulu Swamps, Zambia, Fish. Manage. Ecol. 5, 81–92.
- Triantaphyllou E., Lin C.-T. (1996) Development and evaluation of five fuzzy multiattribute decision-making methods, Int. J. Approx. Reason. 14, 281–310.
- Utne I.B. (2008) Are the smallest fishing vessels the most sustainable? Trade-off analysis of sustainability attributes, Mar. Policy 32, 465–74.
- Vaidya O.S., Kumar S. (2006) Analytic hierarchy process: An overview of applications, Eur. J. Oper. Res. 169, 1–29.
- Van Delft A., Nijkamp P. (1977) Multi-criteria Analysis and Regional Decision-making, Martinus Nijhoff Social Sciences Division, Leiden, Netherlands.
- Van Pelt M.J.F. (1993) Ecologically sustainable development and project appraisal in developing countries, Ecol. Econ. 7, 19–42.
- Vandermeulen H. (1998) The development of marine indicators for coastal zone management, Ocean Coast. Manage. 39, 63–71.
- Varis O. (1989) The analysis of preferences in complex environmental judgements - A focus on the Analytic Hierarchy Process, J. Environ. Manage. 28, 283–294.
- Vogt P.W. (1999) Dictionary of statistics & methodology: a nontechnical guide for the social sciences, Publisher Thousand Oaks: Sage Publications, California.
- Voogd H. (1982) Multicriteria evaluation with mixed qualitative and quantitative data, Environ. Plann. B 9, 221–236.
- Voogd H. (1983) Multicriteria evaluation for urban and regional planning, Pion, London.
- Vreeker R., Nijkamp P. (2001) Advances in multicriteria decision support methods for evaluating scenarios; An application to Thailand, in: A. Research Memorandum 2001-15, Department of Spatial Economics, Free University, Amsterdam.

- Ware S.J., Rees H.L., Boyd S.E., Birchenhough S.N. (2009) Performance of selected indicators in evaluating the consequences of dredged material relocation and marine aggregate extraction, Ecol. Ind. 9, 704–718.
- Wells P.G. (2003) Assessing health of the Bay of Fundy concepts and framework, Mar. Pollut. Bull. 46, 1059–1077.
- Wenger R.B., Rong Y. (1987) Two fuzzy set models for comprehensive environmental decision-making, J. Environ. Manage. 25, 167–180.
- Wheater C.P., Cook P.A. (2000) Using Statistics to Understand the Environment, Routledge (Taylor and Frances Group), New York and London.
- White A.T., Savina G.C. (1987) Community-based marine reserves, a Philippine first, in: Magoon O.T., Converse H., Miner D., Tobin L.T., Clark D., Domurat G. (Eds.), Coastal Zone, 1987, pp. 2022–2036.
- Whitmarsh D.J. (1998) Socio-economic implications of alternative fisheries management strategies. CEMARE Res. Pap. 133.
- Whitmarsh D., Palmieri M.G. (2009) Social acceptability of marine aquaculture: The use of survey-based methods for eliciting public and stakeholder preferences, Mar. Policy 33, 452–457.
- Willet K., Sharda R. (1991) Using the analytic hierarchy process in water resource planning: selection of flood control projects, Socio-Econ. Plann. Sci. 25, 103–112.
- Willmann R. (2000) Integration of sustainability indicators: the contribution of integrated economic and environmental accounting, Mar. Freshwater Res. 51, 501–511.
- Wilson, K. D. P., Leung, A. W. Y., Kennish R. (2002) Restoration of Hong Kong fisheries through deployment of artificial reefs in marine protected areas, ICES J. Mar. Sci. 59, S157–S163.
- Witherell D., Pautzke C., Fluharty D. (2000) An ecosystembased approach for Alaska groundfish fisheries, ICES J. Mar. Sci. 57, 771–777.
- Yahaya J. (1988) Fishery management and regulation in Peninsular Malaysia: Issues and constraints, Mar. Resour. Econ. 5, 83–98.
- Yakowitz D.S. (1998) A multiattribute tool for decision support: Ranking a finite number of alternatives, in: El-Swaify S.A., Yakowitz D.S. (Eds.), Proceedings of the First International Conference on Decision Support Systems (MODSS) for Land, Water, and Environmental Management: Concepts, Approaches, and Applications, Lewis Publishers, Honolulu Hawaii, pp. 205–215.
- Yap H.T. (2000) The case for restoration of tropical coastal ecosystems, Ocean Coast. Manage. 43, 841–851.
- Zeller D., Pauly D. (2005) Good news, bad news: global fisheries discards are declining, but so are total catches, Fish and Fish. 6, 156–159.
- Zeller D., Reinert J. (2004) Modelling spatial closures and fishing effort restrictions in the Faroe Islands marine ecosystem, Ecol. Model. 172, 403–420.
- Zhao Y.W., Yang Z.F. (2009) Integrative fuzzy hierarchical model for river health assessment: A case study of Yong River in Ningbo City, China, Commun. Nonlinear Sci. Numer. Simul. 14, 1729–1736.

# Part III Sociology and Economics

# Farmer Responses to Climate Change and Sustainable Agriculture

Aysha Fleming and Frank Vanclay

#### Contents

1	Intro	duction	283	
2	Disco	ourses	285	
3	Meth	ods	285	
4	Farm	ners' Thoughts About Climate Change	285	
5	Four	Discourses of Climate Change	286	
	5.1	The Discourse of Money	286	
	5.2	The Discourse of Earth	287	
	5.3	The Discourse of Human Responsibility	288	
	5.4	The Discourse of Questioning	288	
6	Discussion			
	6.1	Conceptual Barriers to Climate Change Action.	289	
	6.2	Practical Barriers	290	
	6.3	Information Barriers	290	
7	Орро	ortunities for Change	290	
8	Conc	lusion	291	
Re	ferenc	xes	292	

A. Fleming (S)

Abstract Climate change is a major issue for agricultural sustainability, and changes in farming practices will be necessary both to reduce emissions and to adapt to a changing climate and to new social expectations. A complicating factor is that the processes of behaviour change are complex and can be slow to occur. Discourse analysis is useful in understanding how the discourses farmers are embedded in contribute to resistance to change. Discourses are particular ways of using language in particular situations. They have wide ranging effects on beliefs, values and behaviours. Interviews were conducted in 2008 with 63 respondents, including 22 apple growers, 29 dairy farmers and 12 agricultural consultants in Tasmania, Australia. In undertaking a discourse analysis of the transcripts of these interviews utilising N-Vivo, four specific discourses were identified as being important in shaping farmers' perspectives of climate change and sustainability: Money, Earth, Human responsibility and Questioning. Each discourse contributes to resistance to changing behaviour in particular ways. An understanding of these discourses offers a new approach to facilitating behaviour change.

**Keywords** Climate change • Agriculture • Sustainability • Discourse analysis • Discourses • Behaviour change • Resistance • Barriers to adoption • Tasmania

# **1** Introduction

'What's sustainable? You've got to look at our world as we know it. We're not in a sustainable position at the moment. That's why I say what is sustainable – I don't know.' Interviewee.

Tasmanian Institute of Agricultural Research, University of Tasmania, Private Bag 54, Hobart TAS 7001, Australia e-mail: Aysha.Fleming@utas.edu.au

Sustainability is a concept that is relatively easy to understand but difficult to define in practice. At a simplistic level, it means good environmental management and farming communities that are profitable and prosperous, or put another way, that are viable and vital. But what this means in practice and what specific management practices it infers is open to much debate (Vanclay and Lawrence, 1994, 1995). While sustainability has typically been conceived in terms of land degradation issues, over time sustainable agriculture has had to address a range of other issues including energy use, artificial inputs such as fertilizers and agricultural chemicals, and now climate change.

Climate change is increasingly acknowledged and accepted in science and political arenas. The emerging reality of climate change potentially increases the level of concern about issues of sustainability generally. Many agricultural industries will be impacted by climate change, and agriculture in Australia is projected to be especially affected (ABARE, 2007). Agriculture is likely to face considerable pressure to change its practices to become more sustainable for climate change, both in terms of mitigating emissions of carbon dioxide, nitrous oxide and methane, as well as adapting to the consequences of changes in climate already set in place. The consequences of climate change will be wide ranging, including physical changes to the landscape as well as expected changes in government requirements and market demands. There is a perceived urgency for agriculture to change to mitigate current greenhouse gas emissions and to prepare for future climate scenarios (e.g. CSIRO, 2008; Garnaut, 2008).

Despite the demand for action, few in the agricultural community are actually changing their farming practices because of climate change, at least in Australia (Milne et al., 2008). Potential reasons for inaction are diverse – doubt, complexity, avoidance, and the belief that others are responsible are just some of the likely responses that can limit action. Understanding the causes of this wide variety of responses from a social perspective can reveal new and potentially beneficial insights into behaviour (Potter and Oster, 2008). Therefore, more social research into understanding the factors that limit action for climate change, and how to overcome them, is needed (Trumbo and Shanahan, 2000; Moser and Dilling, 2007).

Climate change occurs on a global scale and over a period that is so long that many people find it difficult to relate to. Thus, the actions needed to influence the climate may be difficult to accept (Flannery, 2005). Further, people have different interactions with climate and understand it to be different things, varying from the expected weather, to the number of extreme events, to the level of carbon dioxide in the atmosphere. Climate change is socially constructed differently in different contexts by different social groups based on the different understandings (Pettenger, 2007). Advocates of change need to be aware of the perceptions of climate change that are prevalent in their particular target groups, and need to be able to deal with a diversity of responses, because the ways in which problems are framed and perceived are crucial factors in determining what solutions are seen to be possible (Irwin, 2001). Yet, when it comes to issues of sustainability and climate change, how farmers' social understandings are constructed is not well understood (Lowe et al., 2006). Knowledge about how social responses are generated in agriculture offers a new perspective in how to create alternative, more positive responses and hence facilitate change (Vanclay, 1992, 2004; Vanclay et al., 2009b).

This paper aims to contribute to the social understanding of climate change by demonstrating how the discourses that all social groups are embedded in are crucially linked to the behaviours that are able to be enacted in that group. Discourses fundamentally shape how all concepts are spoken about, and thought about, and thus able to be acted on (cf. Foucault, 1972). Therefore, discourses provide important knowledge of the forces that shape public perceptions and reveal the processes by which climate change is socially constructed. Awareness of discourse is a practical approach because, if the ways that environmental problems are socially constructed are better understood, a range of solutions can be tailored to fit. This paper aims to link the literature advocating action for climate change with the theory around discourses, and argues for a more socially aware understanding of agriculture and farming. It is hoped that this will offer a new, more successful method for promoting change in farming practices towards sustainable agriculture (see also Fleming and Vanclay, 2009a, b).

#### 2 Discourses

The concept of discourse was introduced in the 1960s by the French philosopher, Michel Foucault. Foucault (1972) maintained that the way language is used has consequences for a whole range of things that go beyond the level of individuals or disciplines, to the very structures of society that shape and limit how people are able to speak, think, and act, and to the social structures that are developed accordingly. Discourses are particular ways of using language in particular situations. They exist at the level of a social group and serve to transmit and construct culture, pass on traditions, question the world, and are fundamentally important in the way we construct our identities.

'We speak with the voices of our communities and to the extent that we have individual voices, we fashion them out of the social voices already available to us, appropriating the words of others to speak a word of our own' (Lemke, 1995, 24–25).

Discourses shape the way we use language. Exposure to particular discourses over time creates perceptions about what is right and wrong, normal or abnormal, and thus significantly shapes how we think and act. This means that discourses are influential social constructions that should be examined, particularly in relation to behaviour change. While the study of discourse is a growing component of many academic fields including environmental studies (Harrison et al., 1996; Darier, 1999; Dryzek, 1997; Hajer, 1995; Carvalho, 2007; Kurz et al., 2005; Bäckstrand and Lövbrand, 2007), the potential for studying discourses as a practical approach to facilitating behaviour change is yet to be realized.

Discourses work toward normalisation and act in opposition to other, competing discourses, and therefore are dynamic and in a state of constant change (Wetherell et al., 2001). However, discourses can be actively changed because the constraints that discourses impose are open to challenge (Darier, 1999). Conflict between discourses creates a point of opportunity for developing new discourses. With an awareness of how a discourse is operating, it becomes possible to conceive how that discourse might be different, or to consider that a different discourse should be adopted, or even to create a new discourse altogether. Therefore, resistance in discourses is a site for agency and transformation. An analysis of resistance in discourses can offer useful insights into behaviour change and it can help to demonstrate the points where new discourses, with new actions and possibilities, might begin.

#### 3 Methods

In 2008, interviews were conducted with 63 individuals from the apple, dairy and agricultural consultant communities in Tasmania, Australia. Sourced through personal contact with industry leaders, the interviews were conducted on-farm, or in-office, taking an average of 40 minutes. The interviews were conducted in a semi-structured manner suitable for recording people's feelings and perceptions. Indicative questions included: What do you think about climate change? What do you think are the causes? What are you doing about climate change on your farm? What do you think should be done by others? What is sustainable agriculture? How are you sustainable? What is your biggest risk? What do you think of the carbon pollution reduction scheme? What else would you like to see the government do?

The questions were intentionally open-ended to allow responses to be freely given and to limit the input (and potential bias) of the interviewer. The interviews were transcribed, entered on an N-Vivo database and examined using a constructivist grounded theory approach (Charmaz, 2006) and a discourse analysis methodology (Wetherell et al., 2001). The discourse analysis involved searching for themes relating to resistance to action for climate change. The analysis of these interviews and a corresponding literature survey are the basis of this paper.

# 4 Farmers' Thoughts About Climate Change

Most interviewees thought that climate change was occurring, and about half believed that they had made personal observations of landscape change, or change on their farm, that they linked to climate change. Yet despite accepting that climate change was occurring, only some thought it was anthropogenic in origin and many were undecided about the cause. Only some believed that it is necessary to mitigate the causes of climate change and/or are willing to do so. They believed that others had more responsibility to act. Some believed that the major responsibility for action lay with government, and that it is pointless for individuals to act without government leadership. Many were concerned about the inclusion of agriculture in the proposed carbon pollution reduction scheme. They were particularly distrustful of various intended government actions which they saw as penalizing farmers.

A small number were confused about the concepts of greenhouse gases, ozone depletion and weather, often conflating these with climate. This group was unable to name any greenhouse gases, and believed the hole in the ozone layer was related to climate change.

Most saw opportunities for Tasmania in a changed climate. They expected Tasmania to be sheltered from the worst effects of climate change and, therefore, compared to the rest of the nation and the rest of the world, to be relatively benefited. More broadly, however, many were worried about the future of the world for their children and grandchildren.

#### 5 Four Discourses of Climate Change

Our analysis of the transcripts through subsequent levels from codes and categories to themes and discourses (Fleming and Vanclay, 2009b; Strauss and Corbin, 1998) resulted in four discourses being identified. Each of the four discourses shows a distinctly different way of talking and thinking about climate change. These are discussed below and, following these descriptions, are contrasted with each other in Table 1.

# 5.1 The Discourse of Money

In the discourse of money, nature is understood as a resource to be monitored, controlled and maximized, and sustainability is about continuing productivity and profit. The main concerns of climate change are about the ongoing viability of business and consistency of action at an international level. Climate change, in the form of a major disruption, is not a concern because physical changes are assumed to be gradual and are expected to be overcome through adaptation. Therefore, how people, governments and other countries act is more important than how the environment changes. There is concern about the equity of actions that might be taken to address climate change. Fairness would demand that everyone acts equally, yet this is not practical as everyone has different capacities and responsibilities for action. The government is not trusted to manage these different capacities and responsibilities effectively, especially in terms of the financial restrictions or taxes they will impose. Finally, individual actions are not accepted as important enough to be contributing to the problem and therefore it is pointless for individuals to act alone.

The discourse of money is characterized by a focus on maximizing profit, maintaining economic growth, supporting technological and financial market fixes, and orchestrating opportunities to maintain competitive advantage into the future. This discourse sees climate change as being able to be overcome through current cultural and social structures, namely market forces and innovation. As solutions to climate change are assumed to be possible, climate change is only perceived as a threat in terms of what decisions are made to inhibit production or penalise agriculture. The solutions are also assumed to be primarily technological and, therefore, agriculture as an industry needs to be especially careful to stay in line with what others do in order to remain competitive. Locally, Tasmania is seen to be particularly sheltered from major environmental impacts, and therefore there may be potential to benefit from climate change, if the only difference is a few degrees increase in temperature.

In the discourse of money, desirable farming practices centre on concerns about effectiveness, efficiency, market relationships, and industry positioning. Farmers' capacities to act for climate change are hindered because costs are perceived as being high, the effectiveness of action unproven, and action is seen as being detrimental to competitive ability. A wait and see approach is favoured, because how others act is crucial for positioning. These views inhibit those in this discourse from acting now, or in supporting Australia to act independently, despite arguments that the costs of inaction are likely to be greater than the costs of action (Garnaut, 2008), that Australia is likely to be particularly affected by climate change, that Australian farmers are particularly likely to suffer this burden financially (ABARE, 2007), and that adaptation is a finite process and unlikely to be sufficient to respond to climate change (Howden et al., 2009).

	Money	Earth	Human responsibility	Questioning
What is nature?	A resource to be monitored, controlled and maximized	A gift to be left untouched and respected	A system that is fragile and needs protection	A system that is infinitely complex, but potentially knowable
What is sustainability?	Continuing productivity and profit	Preserving the purity of nature	Protecting nature for future generations	A better future made possible by ongoing technological and scientific progress
What is climate?	Expected conditions for production	Natural cycles	The experience of weather over time	A scientific understanding based on models and historical records
What is drought?	Exceptional circumstances, business risk	Unpredictable natural event	Reason for better land management	A separate, but compounding issue; climate change is not climate variability
What is climate change?	Future business risk, unfair financial cost in relation to mitigation and adaptation	Natural event	A justification for calling for major change	A complex process, which appears to be taking place. More knowledge needed
What is the cause of climate change?	Multiple causes. Anthropogenic causes often seen to be accelerating a natural trend	Natural process	Anthropogenic is typically accepted but not a defining feature of this discourse	Anthropogenic causes are cautiously accepted, with the complexity of multiple causes emphasised
What is climate change in Tasmania?	Sheltered from extremes, less affected than elsewhere	No different to anywhere else	An opportunity to lead and set an example	Uncertain; current information is inadequate and more detail is needed
What is farming?	A way for earning financial reward through hard work	A life close to nature	Providing an essential service	An application of skill, knowledge and technology
Who has responsibility for solutions to climate change?	Government, corporations, industry bodies, consumers	Gaia, God, the cosmos	All people equally	Future researchers will have a major role when we learn more; current research contributes to the knowledge base
How is science perceived?	Potentially useful, can be complex and may need translation to be practical	Often irrelevant	Science has a role in creating solutions	Science is useful, and skepticism is an important scientific trait
How are global concerns e.g. terrorism, financial crisis, perceived?	Increase business impacts and opportunities	Shows failure to respect nature	Shows need for social transformation	Adds another layer of complexity

Table 1 Comparison of the four discourses that are influencing Tasmanian farmers

# 5.2 The Discourse of Earth

This discourse focuses on the earth and has as a key feature, 'Mother Nature', a divine metaphorical personification that embodies creative and restorative power. While there is concern for the negative effects on the environment that humans are causing, there is a belief that these are sufficiently insignificant to have any real effect and that the world will persist relatively unchanged. In other words, the earth has the power to endure. There is a strong sense that humans do not have dominion over the earth but that the earth has dominion over humans. There is also the sense that the earth is vast and beyond human comprehension.

In the discourse of earth there is a sense that because climate change is controlled by an external force, it might be part of a divine purpose and therefore not of any great concern. In this view, climate change will potentially provoke natural evolution of humans and other species, or humans may be wiped out but the earth will endure albeit in a different state, as occurred with the extinction of the dinosaurs and other major events.

Desirable farming practices in this discourse centre around respect for nature. Farmer's capacities to act for climate change are hindered because humans are not perceived to be able to influence the state of the planet, and the earth has a considerable capacity to withstand change, or homeostatic capacity. Climate change is seen as one aspect of 'the category of environmental insults deriving from industrial society' (Bulkeley, 2000, 319). Equally important problems are degradation, pollution, extinction and the use of environmentally-unfriendly products. While all of these problems are undesirable and even immoral, they are nevertheless not actually able to affect the earth's equilibrium. There is sufficient mystery and trust in the incomprehensible workings of the planet that human attempts to direct the future are naïve and inconsequential.

# 5.3 The Discourse of Human Responsibility

This discourse demonstrates a fundamental difference from the two discourses already discussed because, instead of financial or environmental concerns, it focuses on social action. The discourse of human responsibility is about the necessity of acting for climate change and working together to communicate, collaborate and participate. This discourse has a strong sense of agency and responsibility for action. While it is positive and focused on social action, it is held back by a lack of clear direction in what actions to take.

This discourse is primarily about achieving more public engagement with climate change and about creating more equitable and desirable government policies and even a better world order. However, this is a grand plan and climate change can get lost amidst the focus on transformations of social structures that are demanded. Society is seen to be the problem in this discourse and the tools to change society are identified as being people collaborating and working together, demanding what needs to occur using the power of democracy, yet this process is not actually achieving the major changes required to allow all the actions that are yearned for.

Desirable farming practices in this discourse centre on concerns over the capacity of farmers to meet their responsibility to feed the world's increasing population. The ability of farmers to act for climate change is hindered because actions are inhibited by the need to continually increase output and because of the limitations of current infrastructure, social systems and social norms. Changing consumption patterns and environmental values are seen as being essential in this discourse, but this requires system level transformation, which at this stage is still only being talked about and not yet incorporated into action (see also Harrison et al., 1996).

# 5.4 The Discourse of Questioning

The discourse of questioning is created through the interplay of the hegemonic power science has, which is propagated by the media, but moderated by public opinion. This discourse is focused on issues of fact, truth, knowledge, information and trust. It emphasises aspects of uncertainty or incomplete knowledge, and the complexity of the issue. Vested interests are seen as being likely to exaggerate climate change, and while there is probably some element of truth in how humans cause negative environmental impacts, the extent to which this occurs and how these are best addressed is still unknown and unable to be discerned until the emotional hype has subsided.

The discourse of questioning has doubt and the quest for more knowledge as its key features. Controversial or emotional information is likely to be distrusted and rejected. In this discourse, nothing about climate change is black and white, and everything is arguable and contested. Information is likely to have been found too confusing, too complex, too distant, too tainted, or too difficult to understand. In this discourse, further attempts to engage with finding more information, talking about the issue, or thinking reflexively about it are avoided until such a time as the answer is made sufficiently clear and legitimated by more scientific endeavor. This involves waiting for others to synthesise the information and come up with a position that is generally accepted and supported. At the moment, the most easily adopted positions are either total rejection, or sitting on the fence.

Desirable farming practices in this discourse centre on specialist knowledge and skillful application of technology. Farmers' capacities to act for climate change are hindered because climate change is too uncertain to be actionable, too controversial to be entirely true, and the required changes too radical to be trusted. In this discourse, trust is a particularly important issue, especially trust in whose knowledge and whether that knowledge relates to personal contexts (see also Carolan, 2006).

## 6 Discussion

The general perceptions about climate change of Tasmanian farmers as identified in this research are generally consistent with those found by other studies into public responses to climate change (e.g. Moser and Dilling, 2007; Milne et al., 2008; Lorenzoni et al., 2007; Doulton and Brown, 2009). However, very little research has examined farmers' discourses surrounding climate change for us to compare with our results. This paper seeks to emphasise that responses to climate change are a product of social, rather than individual processes, and therefore more research that takes account of the operation of discourses should be undertaken. We believe that many of the root causes for inaction in the face of climate change are social and discursive. Only a wider account of these social discourses can explain behaviour and thus, resistance.

Many studies of public responses to climate change choose to focus on problems with information, or individual psychology, which are often named the 'barriers' to action (e.g. Bord et al., 2000; Stamm et al., 2000; Stoll-Kleeman et al., 2001; Leiserowitz, 2007; Bostrom and Lashof, 2007; Kollmuss and Agyeman, 2002). While this literature offers important insights into understandings of climate change at an individual level, we believe that there is no such thing as a barrier to change, only legitimate reasons not to change (Vanclay, 1992, 2004). The processes by which these reasons are deemed legitimate or otherwise can be made apparent through analysis of discourses. We argue that a social focus on behaviour change is more useful in relation to facilitating action for climate change than a focus on the specific barriers to change, because it is only through a social approach that the 'practical and discursive constraints of context, both locally and nationally' (Harrison et al., 1996, 215) can be properly addressed.

The many explanations as to why people do not change behaviour that are discussed in the literature can be generally categorized into groupings around conceptual, practical and information barriers. These are summarised below in order to restate our belief that it is not as useful to find out the barriers to action as it is to properly understand the legitimate reasons for inaction. By describing the large number of individual barriers below, we hope to show how overwhelming change can be, if each barrier is to be addressed individually. Instead, we advocate a focus on discourse that offers a more holistic and thus more effective way of understanding and addressing inaction and resistance.

# 6.1 Conceptual Barriers to Climate Change Action

This grouping comprises the many arguments given as to why people can not comprehend climate change due to its complexity. In these arguments, climate is perceived to be a complex science created by multiple interactions between the oceans, land masses and the atmosphere. There are complex effects of climate on the environment, including, but not limited to, the weather. These effects occur over long time scales of years, decades and centuries, so cause and effect connections are difficult to establish and cycles are not often experienced by individuals and/or not accurately remembered. Climate systems and climate cycles are created on a scale that make it seem too distant and too abstract, or too vast and unalterable (Moser and Dilling, 2004). Public understanding of climate change is reliant on science to discover, monitor and potentially solve the problem (Demeritt, 1998), and the media is seen as the conduit for this information transfer from science to the public.

Conflicts between science and the media and the public about truth, values, knowledge, power, responsibility for action, and agency have been the focus of many studies about climate change and environmental sustainability (Potter and Oster, 2008; Boykoff, 2008; Carvalho, 2007; Lorenzoni et al., 2007; Kurz et al., 2005; Sarewitz 2004; Jasanoff, 2004; Clover, 2003; Princen et al., 2002; Dryzek, 1997; Hajer, 1995; Ungar, 1992; Litfin, 1994). While some of this research does use discourse, it is our belief that this has so far been insufficient.

#### 6.2 Practical Barriers

Another category of barriers to change can be conceived as relating to the practical dimensions of the posited solutions. In the conventional diffusion of innovations literature, these include available time, money and social infrastructure, as well as considerations of convenience, ease, flexibility, divisibility, referring to the breakdown of a change in behaviour into the required steps (Rogers, 1983; Vanclay, 1992, 2004; McKenzie-Mohr and Smith, 1999; Pannell et al., 2006). The individual states of motivation, risk, resources, support, individual character traits and skills also play a part. In relation to climate change, Moser and Dilling (2007) have outlined similar barriers to action.

Some scholars advocate a different, more social level approach (Potter and Oster, 2008; Lorenzoni et al., 2007) to change the social structures that limit these behaviours, and to create social mores to normalise the desired behaviours (Griskevicius et al., 2008). We see these as being complementary aspects of the broader concepts of changing discourse. Discourses influence the language used to talk about issues, the types of institutions needed in society and the way these institutions are used (Phillips and Jorgensen, 2002). Therefore, discourses are fundamental in understanding behaviour.

# 6.3 Information Barriers

The final category of barriers is problems of information and its communication. While the critique of the view that the provision of information alone does lead to behaviour change has been well-established since the rise of what is called the 'information deficit model' (see Potter and Oster for a review) some, for example, Sturgis and Allum (2004) still believe that the provision of information will change behaviour. They are not alone in this view as it is widely shared by many scientists.

In the view of those who think information will solve the problem, they see 'information' in simplistic, 'objective' terms, and not in its social context. The barriers to change that these people consider are the lack of information, the lack of access to information, problems in the targeting of information, and the lack of ability of people in understanding the information. There is the view, too, that in a society which is potentially overloaded with information, many people lack the ability to find the information they need, or lack the necessary tools and intellectual resources to evaluate the competing information that is on offer.

In our view, behaviours are embedded within different contexts and situations, and are linked to institutions, social networks and the contexts of place. These must all be considered and adapted or transformed with the adoption of even seemingly simple behaviour changes. We consider that 'problems' or 'barriers' should not be addressed individually, but should be included in the overall account of the reasons for behaviour which we see as being discursive. Changing behaviour on an individual scale is slow and likely to be resisted at many points: 'behavior change is not a one-by-one persuasion task, but a social challenge' (Tribbia, 2007, 248). This is a challenge achievable through working to change the discourses that currently limit behaviour.

#### 7 Opportunities for Change

From the four discourses we have found operating in agricultural circles in Tasmania, there are points of opportunity for change. By understanding the way issues are framed and understood in particular discourses, the ways forward can be framed in a corresponding fashion. This can minimize misunderstandings and tap into existing motivations for action. In this way, an understanding of which discourse is operating in which context can provide a social insight into farmers' characters and positions.

The discourse of money sees resistance to changing practices for climate change as the best way to avoid costs, to focus on other more important problems and to allow time to learn more and therefore increase the likelihood of being more competitive when implementing actions later. However, those who are influenced by this discourse can be motivated to support action by stimulating their need to maintain competitiveness and to be involved with climate solutions developed by industry. To increase the desire for, and adoption of, actions in this discourse, the potential financial benefits and future costs need to be clearly identified. Information about how others are responding, especially at government, industry and consumer levels, is especially important. In this discourse, emphasis on the human responsibility for the environment and emotionally-laden tactics are unlikely to be successful unless tightly connected to issues of financial concern. Therefore, explicitly highlighting the connections between financial problems and climate change, for example reducing input costs by addressing climate change, is of more use than describing other general impacts, no matter how catastrophic, that have financial consequences merely implied.

The discourse of earth resists action for climate change because it sees any action humans can take as being too small and/or irrelevant to make a difference. This discourse sees that other forces are in ultimate control of the earth, and humans cannot affect the outcome of the future path of the planet, or the future of the species. However, this discourse supports action by wanting to promote the value, goodness and wonder of nature and caring for it in a way that is properly respectful and grateful. The best way forward for increasing action in this discourse is by highlighting the multiple environmental benefits of climate change action and emphasising the cultural shift toward sustainability, that is, respect for the earth, that is required. Promotion of the anthropogenic origins of climate change will not be an effective, or necessary, way to promote action in this discourse.

The discourse of human responsibility resists action for climate change because there is confusion about what to do and how to practically implement actions, especially because of concerns about major barriers that are seen to be impossible to fix, like social and governmental structures. However, there is a great deal of willingness to act and therefore significant untapped potential for action is demonstrated in this discourse. If given the resources, those in this discourse are likely to take action up quickly and this is also the most likely discourse to create new forms of action. In this discourse, emphasising the practical information about climate change actions, particularly those involving community or group interaction is the best way forward. Highlighting the reasons for action, or the urgency of action, will not be effective in this discourse as the desire to act is already present and further emphasis can overwhelm. Instead, demonstration of practical ways forward and providing social contacts and the framework for networks will be the most beneficial.

The discourse of questioning resists action for climate change because of distrust or dissatisfaction with information or perceptions of the inability of people to understand or relate to information. This discourse advocates avoidance, denial and delay. However, it has the potential to support action because it accepts that some knowledge is already available for how to proceed. It trusts scientific pursuit to eventually provide the answers and accepts that progress is achieved incrementally, so some action is advisable now. In this discourse, information about the potential benefits of actions needs to be highlighted, rather than information aimed at overcoming scepticism about the causes. In effect, showing this discourse that they too have a vested interest in acting on climate change, and a role to play in producing relevant, 'on-the-ground' knowledge about action for climate change, is the best way forward.

# 8 Conclusion

Each of the discourses identified through our interviews with members of the Tasmanian agricultural community shows a distinctly different way of framing the issues of climate change and sustainability. These are: as an issue of business viability; as an environmental concern; as a call for social action; or as a problem of trust and information. Knowledge of which of these discourses is at work within different social settings allows for different approaches for facilitating behaviour change to be implemented. Each of the discourses provides points of opportunity for action by focusing on the particular aspects that are central to the discourse and that would therefore motivate change by highlighting issues of financial benefit, environmentalism, social action or trusted knowledge respectively.

Climate change means that sustainability is more important than ever, but still understood in a diverse range of ways. Australian agriculture is going to be under significant pressure to implement a wide range of changes in practice for adaptation, mitigation and social responsibility. A social consideration of the agricultural community's behaviours and perspectives is now even more important to consider. More effective approaches to understanding behaviour change are needed because of the urgency of action for climate change. The diversity of social understandings and responses to climate change and sustainability mean that a new method for facilitating change is required. In order to cope with diversity, this new method needs to be focused on a social level of change in order to have a meaningful and significant effect. Awareness of discourse is the most appropriate tool for achieving this level of change.

Acknowledgements Aysha Fleming is supported by a Top-Up Scholarship from CSIRO's Climate Adaptation Flagship and from the Climate Futures for Tasmania project. Ethics approval for the project was granted by Tasmanian Social Sciences Human Research Ethics Committee (H10168). Sincere thanks to the many farmers and agricultural industry spokespersons who agreed to be interviewed, to Claire Hiller for help with the analysis and to Shaun Lisson for commenting on drafts.

# References

- ABARE (2007) Statistical tables, Australian Commodities March quarter, Vol. 14, No. 1, pp. 227–263.
- Bäckstrand K., Lövbrand E. (2007) Climate Governance Beyond 2012: Competing Discourses of Green Governmentality, Ecological Modernization and Civic Environmentalism, in: Pettenger M. (Ed.), The Social Construction of Climate Change: Power, Knowledge, Norms, Discourses, Ashgate, Hampshire, pp. 123–148.
- Bord R., O'Connor R., Fisher A. (2000) In what sense does the public need to understand global climate change? Public Understanding of Science 9, 205–218.
- Bostrom A., Lashof D. (2007) Weather it's climate change? in: Moser S., Dilling L. (Eds.), Creating a Climate for Change: Communicating Climate Change and Facilitating Social Change, Cambridge University Press, Cambridge, pp. 31–43.
- Boykoff M. (2008) Lost in translation? United States television news coverage of anthropogenic climate change from 1995– 2004, Climatic Change 86, 1–11.
- Bulkeley H. (2000) Common knowledge? Public understanding of climate change in Newcastle, Australia, Public Understanding of Science 9, 313–333.
- Carolan M. (2006) Social change and the adoption and adaptation of knowledge claims. Whose truth do you trust in regard to sustainable agriculture? Agric. Human Values 23, 325–339.

- Carvalho A. (2007) Ideological cultures and media discourses on scientific knowledge: re-reading news on climate change, Public Understanding of Science 7, 223–243.
- Charmaz K. (2006) Constructing Grounded Theory: A practical guide through qualitative analysis, SAGE, London.
- Clover D. (2003) Environmental Adult Education: Critique and Creativity in a Globalising World, New Directions for Adult and Continuing Education 99, 5–15.
- CSIRO (2008) An overview of climate change adaptation in the Australian agricultural sector – impacts, options and priorities, Australian Government Department of Land and Water, Canberra.
- Darier E. (1999) Discourses of the Environment, Blackwell, Oxford.
- Demeritt D. (1998) Science, social constructivism and nature, in: Braun B., Castree N. (Eds.), Remaking Reality: Nature at the millennium, Routledge, London, pp. 173–193.
- Doulton H., Brown K. (2009) Ten years to prevent catastrophe? Discourses of climate change and international development in the UK press, Global Environ. Chang. 19, 191–202.
- Dryzek J. (1997) The Politics of the Earth: Environmental discourses, Oxford University Press, Oxford.
- Flannery T. (2005) The Weather Makers: The History and Future Impact of Climate Change, Text Publishing, Melbourne.
- Fleming A., Vanclay F. (2009a) Discourses of Climate Change: Understanding Farmer Resistance, in: Martin J., Winter C. (Eds.), Climate Change Responses across Regional Australia: Social Learning and Adaptation, Latrobe University, Bendigo.
- Fleming A., Vanclay F. (2009b) Using discourse analysis to better inform the practice of extension, Extension Farming Systems Journal (in press).
- Foucault M. (1972) The Archaeology of Knowledge: And The Discourse on Language, trans. A Sheridan Smith, Pantheon Books, New York.
- Garnaut R. (2008) Issue paper 1 Climate change: Land use Agriculture and Forestry, Retrieved 12 June, 2008, from http://www.garnautreview.org.au/CA25734E0016A131/ pages/reports-and-papers.
- Griskevicius V., Cialidini R., Goldstein N. (2008) Social Norms an underestimated and underemployed lever for managing climate change, Int. J. Sustain. Commun. 3, 5–13.
- Hajer M. (1995) The Politics of Environmental Discourse: Ecological Modernization and the Policy Process, Oxford University Press, London.
- Harrison C.M., Burgess J., Filius P. (1996) Rationalising environmental responsibilities: A comparison of lay publics in the UK and the Netherlands, Global Environ. Chang. 6, 215–234.
- Howden M., Nelson R., Crimp S., Park S. (2009) Adapting farming to climate change in the Asia-Pacific region, Greenhouse 2009 conference, Perth.
- Irwin A. (2001) Sociology and the environment, Polity Press, Cambridge.
- Jasanoff S. (2004) States of knowledge: The Co-production of Science and Social Order, Routledge, London.
- Kollmuss A., Agyeman J. (2002) Mind the gap: why do people act environmentally and what are the barriers to proenvironmental behaviour? Environ. Educ. Res. 8, 239–260.

- Kurz T., Donaghue N., Rapley M., Walker I. (2005) The ways that people talk about natural resources: discursive strategies as barriers to environmentally sustainable practices, Brit. J. Soc. Psychol. 44, 603–620.
- Leiserowitz A. (2007) Communicating the risks of global warming: American risk perceptions, affective images, and interpretive communities, in: Moser S., Dilling L. (Eds.), Creating a Climate for Change: Communicating Climate Change and Facilitating Social Change, Cambridge University Press, Cambridge, pp. 44–63.
- Lemke J. (1995) Textual Politics: Discourse and Social Dynamics, Taylor & Francis, London.
- Litfin K. (1994) Ozone Discourses: Science and Politics in Global Environmental Cooperation, Colombia University Press, New York.
- Lorenzoni I., Nicholson-Cole S., Whitmarsh L. (2007) Barriers perceived to engaging with climate change among the UK public and their policy implications, Global Environ. Chang. 17, 445–459.
- Lowe T., Brown K., Suraje D., De Franca Doria M., Haynes K., Vincent K. (2006) Does tomorrow ever come? Disaster narrative and public perceptions of climate change, Public Understanding of Science 15, 435–457.
- McKenzie-Mohr D., Smith W. (1999) Fostering Sustainable Behaviour: An introduction to community-based social marketing, New Society Publishers, Canada.
- Milne M., Stenekes N., Russell J. (2008) Climate Risk and Industry Adaptation, Australian Government Bureau of Rural Sciences, Canberra.
- Moser S., Dilling L. (2004) Making Climate Hot, Environment 46, 32–47.
- Moser S., Dilling L. (2007) Creating a Climate for Change: Communicating Climate Change and Facilitating Social Change, Cambridge University Press, Cambridge.
- Pannell D., Marshall G., Barr N., Curtis A., Vanclay F., Wilkinson R. (2006) Understanding and promoting adoption of conservation practices by rural landholders, Aust. J. Exp. Agr. 46, 1407–1424.
- Pettenger M. (2007) The Social Construction of Climate Change: Power, Knoweldge, Norms, Discourses, Ashgate, Hampshire.
- Phillips L., Jorgensen M. (2002) Discourse Analysis: as theory and method, Sage, London.
- Potter E., Oster C. (2008) Communicating climate change: public responsiveness and matters of concern, Media International Australia 127, 116–126.
- Princen T., Maniates M., Conca K. (2002) Confronting Consumption, MIT Press, Cambridge.

- Rogers E. (1983) Diffusion of Innovations, 3rd ed., Macmillan, New York.
- Sarewitz D. (2004) How science makes environmental controversies worse, Environ. Sci. Policy 7, 385–403.
- Stamm K., Clarke E., Reynolds E. (2000) Mass communication and public understanding of environmental problems: the case of global warming, Public Understanding of Science 9, 219–237.
- Stoll-Kleeman S., O'Riordan T., Jaeger C. (2001) The Psychology of Denial concerning climate mitigation measures: evidence from Swiss focus groups, Global Environ. Chang. 11, 101–117.
- Strauss A., Corbin J. (1998) Basics of qualitative research: techniques and procedures for developing grounded theory, Sage publications, Thousand Oaks.
- Sturgis P., Allum N. (2004), Science in society: re-evaluating the deficit model of public attitudes, Public Understanding of Science 13, 55–74.
- Tribbia J. (2007) Stuck in the slow lane of behaviour change? A not-so-superhuman perspective on getting out of our cars, in: Moser S., Dilling L. (Eds.), Creating a Climate for Change: Communicating Climate Change and Facilitating Social Change, Cambridge University Press, Cambridge, pp. 237–250.
- Trumbo C., Shanahan J. (2000) Social research on climate change: where we have been, where we are and where we might go, Public Understanding of Science 9, 199–204.
- Ungar S. (1992) The rise and (relative) decline of global warming as a social problem, The Sociological Quarterly 33, 483–501.
- Vanclay F. (1992) Barriers to adoption: a general overview of the issues, Rural Society 2, 10–12.
- Vanclay F. (2004) Social principles for agricultural extension to assist in the promotion of natural resource management, Aust. J. Exp. Agr. 44, 213–223.
- Vanclay F., Lawrence G. (1994) Farmer Rationality and the adoption of environmentally sound practices: a critique of the assumptions of traditional agricultural extension, Eur. J. Agric. Education and Extension 1, 59–90.
- Vanclay F., Lawrence G. (1995) The Environmental Imperative: Ecosocial Concerns for Australian Agriculture, Central Queensland University Press, Rockhampton.
- Vanclay F., Leith P., Fleming A. (2009) Understanding farming community concerns about adapting to a changed climate, in: Filho W., Mannke F. (Eds.), Interdisciplinary Aspects of Climate Change, Peter Lang, Frankfurt am Main: Germany, pp. 229–244.
- Wetherell M., Taylor S., Yates S. (2001) Discourse as data: a guide for analysis, Sage, London.

# The Use of the *Marasha* Ard Plough for Conservation Agriculture in Northern Ethiopia

Jan Nyssen, Bram Govaerts, Tesfay Araya, Wim M. Cornelis, Hans Bauer, Mitiku Haile, Ken Sayre, and Jozef Deckers

#### Contents

1	Intro	duction	296
2	Tilla Ethic	ge in the Agricultural System of the Northern opian Highlands	297
3	Intro	duction of Implements	299
4	Indig	genous Use of the Marasha for In Situ Soil and	
	Wate	r Conservation	299
	4.1	The <i>Terwah</i> Technique	299
	4.2	Shilshalo	300
	4.3	Derdero	302
5	Bed-	and-furrows for the Northern Ethiopian	
	High	lands	303
6	Futu	re of the <i>Derdero</i> + System	304
7	Conc	lusions	306
Re	ferenc	es	307

J. Nyssen (🖂)

Ghent University, Department of Geography, 9000 Gent, Belgium

e-mail: jan.nyssen@ugent.be

Abstract Indigenous tillage systems are often undervalued in conservation agriculture (CA). In Ethiopia, since the 1970s there have been several attempts to develop and implement often major modifications to the marasha, the traditional ox-drawn ard plough, with the main aim of creating various types of surface depressions. The establishment of furrows and ridges increases soil moisture and grain yield and reduces soil loss. Dissemination of the modified tools, however, remains limited. Recent tendencies are towards testing relatively simple conservation agriculture tools. Major challenges remain, however; the need for capacity building and problems in marketing the tools. From experimental plots, often worked with exotic tools, there is a long road to real adoption by farmers. Rather than developing yet another CA tool, we investigate whether CA-based resource-conserving technologies might be achieved successfully with simple changes to the use of the marasha. On-farm observations on traditional conservation techniques were carried out throughout the northern Ethiopian highlands, and experiments were conducted involving resource-conserving technologies. Farmers traditionally use the marasha and plough for various types of in situ soil and water conservation by creating surface depressions, either at the moment of sowing (terwah, derdero) or after crop emergence (shilshalo). Building upon this indigenous knowledge, we further developed resource-conserving technologies into a system named derdero+, whereby the traditional ard plough was found suitable for a "bed-and-furrow" system. From the socio-economic point of view, implementation of permanent beds and retention of stubble leads to decreased oxen (and straw) requirements, but also to an increased need for weeding in the first years.

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_16,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Nyssen et al., Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2010014

To overcome that problem, we introduced glyphosate herbicide into the tillage system. The decreased runoff (-51%) and soil loss (-81%) allow protection of the downslope areas from flooding, but soil nutrient build-up and soil structure improvement are slow processes, and hence the full benefit of the permanent bed system can only be expected after some years. Overall, this type of resource-conserving technology can be part of the ongoing intensification process which includes physical soil and water conservation, slope reforestation and irrigation development. It has, however, its own niche: the cropped land sensu stricto, i.e. the most important part of the land, both for the farmer and for a nation that is striving for long-term food security.

**Keywords** Animal traction • Ard plough • Conservation tillage • Conservation agriculture • *Derdero*+ • *Marasha* • Indigenous knowledge • Permanent beds • Raised beds • Resource-conserving technology • Soil and water conservation

#### **1** Introduction

In the northern Ethiopian highlands, smallholder agriculture is largely dominant and tillage takes place with a pair of oxen pulling the ard plough or *marasha* (Simoons, 1958; Nyssen et al., 2000; Solomon et al., 2006) (Fig. 1), a tool similar to that of the eastern Mediterranean and large parts of the Arab world

J. Nyssen et al.

(Audebeau and Mosséri, 1916; Varisco, 1982, 2004). The main conservation approach involves terracing by stone bunds, which is very efficient for soil and water retention within the plot. Crop yield and farmers' income increased on average by 10% (Desta et al., 2005). However, the effect of the stone bunds on crop production is concentrated along the bunds (Fig. 2) (Vancampenhout et al., 2006). Agricultural land surface management, and particularly the creation of minor closed depressions, is needed to have a conservation effect over the whole farmland. This fits into a conservation agriculture (CA) approach to enhance production, preserve natural resources and reduce poverty (Aune et al., 2001), through (a) in situ conservation of moisture, reducing runoff and evaporation; (b) reductions in erosion within farmland; (c) crop management practices that meet the above goals and increase labour efficiency and reduce the oxen requirement, leading to reductions in the demand for crop residues as livestock feed, and (d) enhanced soil fertility, to make efficient use of increased retention of rain.

Conservation agriculture (CA) is based on 3 principles (Bradford and Peterson, 2000; Verhulst et al., 2010): (1) reduction in tillage – the objective is to achieve zero tillage, but the system may involve controlled tillage seeding systems that normally do not disturb more than 20–25% of the soil surface; (2) retention of adequate levels of crop residues and surface cover on the soil surface to protect it from water and wind erosion, to reduce water runoff and evaporation, to improve water productivity and to enhance soil physical, chemical and biological properties associated



**Fig. 1** The first ever photograph of an Ethiopian farmer with his oxen and *marasha*, near Senkata (Tigray) in March 1868 (ⓒ Royal Engineers of the British Army, reprinted with

permission of the King's Own Museum, Lancaster, UK) and detail of the metal tine with wooden "ears" (Photo Leigh Vial)



**Fig. 2** A sorghum and maize farm in the May Zegzeg catchment, Tigray, October 2005. The implementation of stone bunds for soil and water conservation has led to strongly improved yields in the vicinity of the bunds (arrow shows strong maize growth). In the foreground, *terwah* furrows remain visible in the harvested land

with long-term sustainable productivity, and (3) use of crop rotations for the beneficial effects of some crops on soil conditions and on the productivity of subsequent crops, and to mitigate possible weed, disease and pest problems related to crop residue retention. These CA principles are applicable to a wide range of crop production systems from low-yielding, dry, rainfed conditions to high-yielding, irrigated conditions. However, the techniques to apply the principles of CA will be very different in different situations, and will vary with biophysical and system management conditions and farmer circumstances (Verhulst et al., 2010). Applying CA also means altering generations of traditional farming practices and implement use. As such, the movement towards CA-based technologies is normally comprised of a sequence of step-wise changes in cropping system management to improve productivity and sustainability, implementing resource-conserving technologies in line with the CA principles.

Particularly, bed planting provides an opportunity to reduce compaction by favouring traffic to the furrow bottoms. The next logical step to increase the sustainability of beds is to make them permanent, reducing or avoiding tillage and retaining and distributing crop residues on the surface, reshaping the beds only as needed (Sayre, 2004). Introducing permanent beds, reduced tillage and residue preservation is expected to conserve water and as such will allow sowing early and cultivating crops and varieties with a longer growing period.

In an experimental setup with conservation agriculture-based resource conservation technologies in the North Wollo region of Ethiopia, McHugh et al. (2007) found that ridges significantly increased soil moisture and grain yield and reduced soil loss, whereas no-till minimised soil loss, but reduced yield. Yet, from experimental plots, often worked with exotic tools, there is a long road to real adoption by farmers. Leeuwis and Van den Ban (2004) demonstrated that innovations in farming may only spread spontaneously when they are embedded within the local socio-economic and cultural context. Rajaram et al. (1991) further mention that indigenous tillage systems are often undervalued. Hence, rather than developing yet another CA tool, we investigate here whether CA-based resource-conserving technologies might be achieved successfully with simple changes to the use of the marasha. Simplicity is a key concept for CA-based resource-conserving technologies. Hence our working hypothesis has been that further adaptations of the marasha can and will be made, but wholesale changes are probably unnecessary to the ox-drawn implement. After all, the farmers of Ethiopia have been adjusting this tool for thousands of years so as to meet their requirements under Ethiopian soil conditions.

We have chosen a step-by-step, empirical approach that involves intimate farmer participation throughout the initial research on possible strategies/technologies, the testing and modification of the most relevant possibilities, and finally, the extension of the final products.

# 2 Tillage in the Agricultural System of the Northern Ethiopian Highlands

The tropical highlands of Ethiopia are densely populated and have been cropped intensively for centuries (Nyssen et al., 2004). Agricultural sustainability problems related to soil erosion and fertility decline have arisen throughout this agro-ecological zone. The agricultural system is under stress due to the shrinking cultivated area per household, reduced fodder availability and land degradation (Aune et al., 2001).

This study concerns the Tigray and northern Amhara (Lasta) regions in the northern Ethiopian



Fig. 3 The target area

highlands (Fig. 3). The main rainy season (>80% of total rainfall) extends from June to September but is sometimes preceded by three months of dispersed less intense rains. Average yearly precipitation is 500-800 mm. Agriculture in the northern Ethiopian highlands consists exclusively of smallholder farming activities, with landholdings approximately equal in size. Each household crops two or three parcels of land, with a combined area of between 0.5 and 1 ha (Nyssen et al., 2008). The agricultural system in the northern Ethiopian highlands has been characterised as a "grain-plough complex" with annually cropped fields. The main crops are barley (Hordeum vulgare), wheat (Triticum sp.), sorghum (Sorghum bicolor), maize (Zea *mays*) and tef (*Eragrostis tef*), an endemic cereal crop. Different species of pulses are also an important part of the crop rotation (Nyssen et al., 2008). As rainfall is inadequate and unpredictable, crop production is threatened by chronic soil moisture stress. Precipitation is often intense, leading to high runoff and temporary waterlogging. Cereal grain yields are low

 $(<2 \text{ t ha}^{-1})$ . Moreover, crops are often weedy and nitrogen-deficient, soil structure is poor, and soil erosion is widespread (Tadesse et al., 1996; Asefa et al., 2004; Spaccini et al., 2006; Nyssen et al., 2008).

Soil tillage with the *marasha* is commonly used to prepare flat seedbeds. The tool, pulled by a pair of oxen, breaks, but does not turn, the soil to provide weed-free seedbeds at sowing and to enhance infiltration (Melesse et al., 2008). The farmers adjust the tillage depth by modifying the length of the ploughshare and its angle and by the downward pressure on the handle. The amount of land a pair of oxen can plough in one day varies with oxen strength and with soil type and is estimated at about 0.25 ha day<sup>-1</sup>. Multiple tillage operations are carried out in order to prepare a fine seedbed, particularly for tef, with 3-5 operations. This leads to poor physical soil properties, as expressed by very hard topsoils and difficult tillage conditions when the next crop needs to be sown. After harvesting, stubble grazing is widespread, followed by several tillage operations.

Oxen are of vital importance for assuring livelihood security as they can be sold when the household experiences a stress situation such as drought. There is also a high social prestige associated with ox ownership (Aune et al., 2001). The productivity of the Ethiopian livestock system is declining as an increasing part of the biomass resources are consumed by oxen and other livestock. A change in the livestock population and productivity will not occur unless the tillage system is modified and overall crop productivity improved.

Two sites, both on Vertisols, were selected for experimental studies on land preparation techniques (Fig. 3): (1) the May Zegzeg catchment near Hagere Selam (2600 m a.s.l.), where catchment management including zero grazing has been implemented since January 2004 (Nyssen et al., 2009) and (2) the Gum Salasa area near Adi Gudom (2090 m a.s.l.) (Tewodros et al., 2009; Tesfay et al., 2010; Tigist et al., 2010). Vertisols are challenging for land preparation: they are very hard when dry, and in the rainy season, when their upper part gets moistened, clay swelling leads to soil sealing, poor infiltration and waterlogging. Due to its resistance to waterlogging, tef is a preferred crop on this soil type, but the repeated ploughing that is done to prepare its fine seedbed leads to collapse of the soil structure and hence leaves behind a compacted topsoil.

#### **3** Introduction of Implements

In the past, researchers and organisations repeatedly attempted to replace or modify the marasha plough, for CA operations among others. The typical trajectory of adoption is similar to that of the steel mouldboard ploughs introduced by the Italians in the 1930s (Fig. 4). This was unsuccessful at smallholder level due to its heavy weight, the need for complicated adjustments and the higher power requirements than those of the Ethiopian ard, especially in soils with high clay contents (Solomon et al., 2006). Since the 1970s there have been several attempts to develop and implement a broad bed and furrow maker in Ethiopia (Jutzi et al., 1987). However, like other implements, dissemination of broad bed makers was very limited (Solomon et al., 2006). Recent tendencies are towards testing relatively simple CA-based resource-conserving technology tools (Melesse et al., 2009). Major challenges remain, however; the need for capacity building and problems in marketing the tools (Rockström et al., 2009).

On the other hand, research on the raised bed and furrow making in central Ethiopia mainly focused on draining excess water for the optimum production in Vertisol-dominated areas where soil water is generally not a limiting factor (Abiye et al., 2002).

As preparing beds and furrows with hand tools is not an option, except sometimes in irrigated land, our research has been oriented towards the use of the nonmodified *marasha* for land preparation in a raised-bed CA-based resource conservation system.

# 4 Indigenous Use of the Marasha for In Situ Soil and Water Conservation

On-farm observations were carried out at various locations throughout the Tigray and Lasta highlands, and interviews with farmers and agricultural extension workers allowed characterising the prevailing in situ soil and water conservation techniques using the *marasha*: *shilshalo*, *terwah* and *derdero* (Table 1).

#### 4.1 The Terwah Technique

The terwah technique, a traditional tillage system (Table 1; Fig. 2), is especially used in tef, where the farmers prepare furrows on the contour at regular intervals (approx. 1.5 m) to conserve water. This technique differs from the guide furrows which one can see in many fields, by the fact that (1) the furrows are deeper and wider, and (2) the marasha is adjusted to prepare these furrows, whereby specially prepared doogri (wings) are fitted on it. As a consequence, part of the runoff is trapped in the terwah and allowed to infiltrate. At these places the crop will grow particularly well. This technique is applied especially in tef, because this crop is sown on the surface after seedbed preparation without more ploughing (Deckers et al., 2001), allowing the farmer to also sow inside the terwah, which avoids the perception that the terwah is occupying valuable land. Using the terwah system for CA-based resource-conserving technologies with other



**Fig. 4** Italian mouldboard ploughs imported into Ethiopia (near Mekelle, in 1938, photo Guidotti). "Gift of H.E. the Head of State to the inhabitants of Tigray" states the original legend of this photograph obtained from the Istituto Agronomico per

l'Oltremare (Florence, Italy). As for other, more recent metal tools, the technology was not adopted by the Ethiopian smallholder farmers, especially due to excessive draught requirements and problems in maintenance

		Tillage system	
	Moment of implementation	Broad-spaced, deeper furrows on contour	Narrow-spaced, deeper furrows on contour
Traditional in situ conservation tillage	Yearly, after emergence Yearly, at last tillage (at sowing)	<i>Terwah</i> (on tef only) (Sect. 4.1: Fig. 2)	Shilshalo (on sorghum and maize only) (Sect. 4.2; Fig. 5) Derdero (on all crops) (Sect. 4.3; Fig. 6)
Modified resource- conserving tillage	Yearly, at last tillage (at sowing) Permanent; yearly, refreshed at sowing	<i>Terwah</i> , retaining stubble (on all crops) (5.3; Fig. 9)	Semi-permanent beds = <i>derdero</i> , retaining stubble (5.2; Fig. 8) <i>Derdero</i> + = permanent beds, retaining stubble and using non-selective herbicides (before furrow refreshment and sowing) (5.1; Fig. 7)

**Table 1** Overview of traditional in situ soil and water conservation in Ethiopia and modified conservation agriculture-based resource-conserving technologies, all using the *marasha* 

crops than tef exists as a concept in Ethiopia but meets resistance. A senior agricultural officer in Tigray explained that agricultural experts prefer to focus on complicated implements such as the tied ridger while neglecting traditional techniques.

The terwah technique differs from michdad or feses, which are graded furrows to evacuate excess water from waterlogged areas or from whole farm fields (Erkossa et al., 2005), especially in high rainfall areas where it may lead to rill erosion (Woldeamlak and Sterk, 2003). However, in most of northern Ethiopia, moisture is an important limiting factor for agricultural production, hence in situ moisture retention is important. The improvement in crop performance in the vicinity of the bunds and terraces constructed along the contour (Vancampenhout et al., 2006) is a good indicator for the need to make contour furrows in the farm fields. An experimental site equipped with lined runoff and sediment collectors (Tewodros et al., 2009; Tesfay et al., 2010; Tigist et al., 2010) was used in 2005, 2006 and 2007 to compare the practice of ploughing and broadcast sowing on plain seedbeds with the *terwah* system, among other things. Here, the technique was shown to be effective for in situ soil and water conservation, reducing runoff on average by 11% and soil loss by 36% (Table 2). An important result of these runoff plot measurements, however, is that the terwah technique is efficient especially at the beginning of the rainy season (when soil moisture is most needed by the plants), whereas during the second part of the rainy season the furrows get filled with sediment;

the land behaves from then onwards in a similar way to the plain tilled land (Tewodros et al., 2009).

Currently, the *terwah* method is promoted by the Agricultural Office extension services, who additionally advise lifting the plough from time to time (every 4 or 5 m) when preparing the *terwah*. In this way, the furrow is interrupted like a tied ridge; hence lateral drainage is avoided or slowed down.

#### 4.2 Shilshalo

Throughout Ethiopia, when farmers grow sorghum or maize, they use the marasha to create contour furrows within the standing crop during the (second) weeding operation (Gebreyesus et al., 2006). The technique has also been documented in Yemen (Bédoucha, 1986; Varisco, 2004). Besides weeding and plant thinning, the aim of the practice (miguemas in Tigrinya, shilshalo in Amharic and south Tigray; Table 1) is to enhance runoff capture, particularly in semi-arid areas. In our target area, and despite the existence of very heavy rains in August when shilshalo is practised (Nyssen et al., 2005), the furrows are made along the contour and are slightly curved upwards at both ends of the farmland to enhance the water harvesting effect. In addition, plant physiologists have pointed to the sorghum crop root pruning that takes place and which enhances root growth (Blum et al., 1977; Rajaram et al., 1991; Blum, 2004).

			Runoff											Soil los	s							
			Conventic	mal		Derdero+				Terwah				Conven	tional	Derdero	+		Terwah			
Site	Year	и	${\rm m}^3~{\rm ha}^{-1}$	s.d.	RC	m <sup>3</sup> ha <sup>-1</sup>	s.d.	RC	%	m <sup>3</sup> ha <sup>-1</sup>	s.d.	RC	$\mathcal{O}_{\mathcal{O}}$	t ha <sup>-1</sup>	s.d.	t ha <sup>-1</sup>	s.d.	$_{0}^{\prime \prime }$	t ha <sup>-1</sup>	s.d.	%	Source
Gum Selasa	2005	7	647		0.15	255		0.06	40	377		0.09	60	19.5		4.6		23	7.6		39	(Tewodros
Gum Selasa	2006	7	928	152	0.27	467	104	0.13	50	782	60	0.22	81	9.7	2.3	1.4	0.3	14	4.6	1.6	47	et al., 2009) (Tigist et al.,
Gum Selasa	2007	7	1158	224	0.27	650	444	0.15	56	1366	229	0.32	118		I	I	I	I	I	I	I	(Tesfay et al.,
May Zegzeg	2005	с	1620		0.16	1496		0.14	92	1592		0.15	98	10.4		8.2		78	11.0		107	2010) (Fekadu 2006)
May Zegzeg	2007	с	1311	149	0.35	920	126	0.24	70	1129	326	0.30	86		I	I	I	Т	I	I	I	(Own data)
Average			1133		0.24	758		0.14	62	1049		0.22	89	13.2		4.7		38	7.7		64	

onal	
aditio	
ith tr	
w be	
npar	
IS COI	
de) a	
rbici	
ut he	
witho	
oth ,	
ng (b	
idguc	
<i>th</i> plc	
terwc	
and	
ro+	
derde	
with .	
ated	
s tre	
rtisol	
on Ve	
and c	
arml	
rom 1	
er) fi	
temb	
/-Sep	
(Jul)	
loss	
l soil	
ff and	
Runo	ent
2	gem



**Fig. 5** *Shilshalo* ploughing as practised on sorghum, some weeks after emergence in the May Zegzeg catchment. In one tillage operation, weeding, thinning and root pruning is done; the furrows also decrease runoff and enhance infiltration



**Fig. 6** The indigenous *derdero* technique applied on a farmer's field with fenugreek near Densa (Lasta), August 2007. Book for scale. Tillage after sowing has created the typical bed-and-furrow topography and brought most of the seeds to the beds

In May Zegzeg, except in three farm fields, *shilshalo* was practised by all farmers who planted sorghum (Fig. 5). A comparison of crop yields sampled on 4-m<sup>2</sup> plots in these three farmers' fields yielded no significant difference with 14 randomly selected farm plots where *shilshalo* was applied: 0.86  $(\pm 0.34)$  t ha<sup>-1</sup> grain on plots with *shilshalo* against 0.90  $(\pm 0.56)$  t ha<sup>-1</sup> without *shilshalo*. Yet, all farmers in that area claim that without implementing *shilshalo* one can hardly grow sorghum.

Gebreyesus et al. (2006) obtained on an experimental plot in Avergelle, 100 km away, but lower, warmer and less rainy, mean sorghum yields of 1.78 t ha<sup>-1</sup> for *shilshalo* and 1.48 t ha<sup>-1</sup> for crops where no furrows were created after plant emergence (results of two years on an experimental plot with three replicates). This tends to indicate that the shilshalo practice is especially essential for sorghum production in rain-deficit areas. McHugh et al. (2007) also mention the indigenous practice of shilshalo which is practised by many farmers in Wello on their sorghum crops for the sake of moisture conservation. Tilahun et al. (2002) further recommended sowing an intercropped legume at the moment of shilshalo. Some authors, however, merely recognise shilshalo as a weeding technique (Rezene et al., 1992; Hussein et al., 2000). While identifying the shilshalo technique, Sasakawa Global 2000 abandoned it on its CA experimental plots with maize, replacing it by glyphosate spraying (Ito et al., 2007).

#### 4.3 Derdero

In Lasta, the derdero system is applied by many farmers, especially for fenugreek (Trigonella foenumgraecum), wheat and tef. At the last tillage operation, the farmer broadcasts the seeds over the surface, and then prepares beds and furrows along the contour using the *marasha*, in order to conserve water (Table 1). Soil and seeds that are on the position of the furrows are moved to an upper position on the beds in the ploughing process. Most of the plants will then germinate on the beds (Fig. 6). The derdero technique is especially used on Vertisols: in rain-rich locations or rain-rich years, it allows one to have the plants grouped in a slightly higher position on the ridges, i.e. protected from waterlogging, while the excess water is drained towards the furrows where it ponds and slowly infiltrates.

Whereas *shilshalo* is applied after emergence (particularly of sorghum and maize), *derdero* tillage is done at sowing time, so that surface depressions exist from the date of sowing onwards.

This tillage system shows many similarities to the "corrugaciones" tillage system in wheat-growing areas in North America. Farmers sow the field, incorporate the seed with a last ploughing operation and subsequently make furrows. In recent years, the "corrugaciones" are frequently used in rainfed systems, where the furrows either serve to drain excess water of heavy rain showers, or to avoid runoff, and as such enhance the water harvest capacity of the field in semi-arid areas (Medina et al., 2003). A similar tillage system has been reported in Yemen (Behnstedt, 1987; Varisco, 2004).

# 5 Bed-and-furrows for the Northern Ethiopian Highlands

Having noted the appropriateness of the *marasha* for surface shaping for conservation, such as *terwah*, *shilshalo* and *derdero*, the next logical step was then to link the tool up with the wide international body of knowledge on CA (Wall, 2007). Through discussion, empirical design, formal experiments and promotion of on-farm use, our current concept of CA and CA-based resource-conserving technologies for smallholder farmers in the north Ethiopian highlands came to include the following:

- limited tillage activity, to enhance soil structure and to decrease the cost of maintaining oxen;
- use of *marasha* as a tool for CA-based resourceconserving technologies;
- leaving half the straw length as standing stubble after harvest;
- crop rotation;
- critical use of non-selective herbicide just before sowing; and
- abandonment of stubble grazing (to improve soil fertility, to decrease compaction and to avoid

browsing of standing stubble as well as roaming livestock just after spraying of herbicide).

This newly developed tillage system is called derdero+, where the "plus" stands for the improvements made. The basic system that has been developed consists of permanent beds with standing stubble (5.1 in Table 1). Here, parallel furrows are made on the contour at intervals of approx. 60 cm, using half the width of the yoke as a measurement (Fig. 7). In the first year of implementation, the furrows are refreshed a few times during the period of land preparation. At sowing, crops are broadcast over the land, and the furrows reshaped, whereby the soil is moved to the beds, thereby covering the seed. In subsequent years, one single tillage pass at the moment of broadcast sowing allows a refreshment of the furrow, reshaping of the bed and covering of seeds at the same time (Fig. 7).

The *derdero+* land preparation system (Figs. 7 and 8), in which the farmer needs to adjust his *marasha* in such a way that the sediment uplifted from the furrows is well spread over the beds, has proven to work with wheat (Tesfay et al., 2010), barley, chickpea, maize and grasspea. For tef, the implementation is slightly different: on the day of sowing, the furrows are first refreshed by *marasha*, whereafter the seed is sown over the surface, including beds and furrows. The conventional incorporation of the seeds into the top 1 cm is done using thorny *Acacia* sp. branches.



**Fig. 7** The *derdero*+ system involves shaping of beds and furrows by a pair of draught animals with attached *marasha* (on a farmer's field in May Zegzeg, July 2009). Only a single, broad-spaced tillage operation is needed. The farmer replaced one of the oxen in the span by a (weak) cow, as only

the sediment accumulated in the previous year's furrows needed to be reworked and less power was needed. Note how the left ox walks in the furrow, which provides guidance for the position of the plough tine in the immediately upslope furrow



**Fig. 8** Wheat on permanent beds (*derdero+* system) in a farmer's field in May Zegzeg, August 2007. Beds were reshaped in a single tillage operation immediately after broadcast sowing

Results from our experimental sites show that this system leads to an average decrease in runoff of 38% and in soil loss of 62% (Table 2).

#### 6 Future of the Derdero+ System

One particular problem that had to be overcome was weed infestation, as frequent tillage no longer takes place: at the beginning of our experimentation, there were large invasions of *Cynodon dactylon*, which could not be controlled by hand weeding and which particularly affected tef growth (Tigist et al., 2010) (Fig. 9).

We have recently introduced the use of glyphosate herbicide on the permanent beds, a few days before sowing, when maximum emergence of the seed bank has taken place. There is a good rationale for the use of herbicides, since tillage operations are carried out only by men and weeding is done mostly by women and children. Hence the introduction of derdero+ without use of herbicides would have led to significant shifts in workload from men to women and children. However, follow-up of appropriate use and dosing of the herbicide is of utmost importance (Christoffoleti et al., 2008). The constraints related to herbicide weed control are related to the fact that spraying is a knowledge-intensive technology. Hence, the extension of CA-based resource-conserving technologies needs to be combined with innovation systems that support training and knowledge transfers. As used to be the case in Ethiopia for fertiliser not so long ago, there is no structure/system in place at the moment for such inputs. Our working example is deemed to catalyse the innovation network and make the needed change to input and output markets.

Two intermediate systems were also introduced, which can be very useful for farmers who hesitate to use herbicide or who believe that "a good farmer ploughs his land many times" (Melesse et al., 2008). Firstly, to manage semi-permanent beds we apply the *derdero* technique, whereby in addition we keep standing stubble on the land (5.2 in Table 1). After several normal tillage operations, without herbicide application, the land will be shaped into contour beds and furrows immediately after broadcast sowing and beds and furrows remain throughout the crop-growing period. The second intermediate system consists of *terwah* ploughing for all crops, as promoted by the Bureau of Agriculture (5.3 in Table 1); traditionally it used only to be done for tef (Fig. 10).

At this initial stage of implementation, our agronomic results for permanent beds do not show a significant difference in crop and straw yield with traditional tillage. We measured a slight increase in wheat yield (Tewodros et al., 2009; Tesfay et al., 2010) and a decrease in tef yield (Tigist et al., 2010) (Fig. 9), which was most probably related to the earlier mentioned weed problems in the absence of herbicide use. Further increase is expected to occur in the future, when the weed seed bank becomes exhausted and soil physical and chemical properties improve. Obviously, for the same crop yield, labour investment and draught force requirement are strongly reduced with derdero+, with only one tillage pass that consists of simply refreshing furrows (Fig. 7). The use of cows and donkeys for tillage becomes possible. Similar to our case, Wall (2007) found that in many developing countries, one of the most important benefits of CA for



**Fig. 9** Average grain yield (kg/ha) trend in conservation and conventional agriculture in Gum Selasa. Different letters stand for significant difference (P < 0.05) between treatments

per year. D + = derdero + (without use of herbicide), TER = terwah, CON = conventional tillage (after Tesfay et al., 2010)



**Fig. 10** The stripped pattern of this wheat farm at the ripening stage in May Mereb (5 km west of Hagere Selam; October 2005) is due to enhanced soil moisture in the area around *terwah* furrows (see detail in Fig. 2). Traditionally, *terwah* ploughing is only used with tef, but in recent years the local Bureau of Agriculture has promoted the establishment of such broadspaced deeper furrows at sowing time of other crops as well

smallholder farmers is the reduction in labour requirements, both by oxen and humans, for tillage. However, if only manual weeding is practised, then labour requirements may increase, making the system unattractive to farmers (Muliokela et al., 2001). Hence the use of glyphosate in our *derdero*+ CA system.

The abandonment of stubble grazing, which has definite advantages in terms of soil conservation (Nyssen et al., 2009), is a prerequisite for implementation of resource-conserving technologies. The issue of leaving standing stubble fits in the same paradigm as the establishment of exclosures (Descheemaeker et al., 2006; Wolde et al., 2007), where previously overgrazed rangeland has been left to regrow and fodder is taken through cut and carry. The only way to get out of the vicious circle of land degradation caused by overgrazing is to increase total biomass production per land unit, so that one can cut and carry part of the straw and leave part for system sustainability. The aimed-for increase in productivity would allow both harvesting the upper part of the straw and leaving tall standing stubble. Zero grazing is promoted in the target area but meets with resistance as there is a strong tradition of offseason grazing. In May Zegzeg, where free grazing has been abolished, the developed *derdero*+technology has been implemented on ten farmers' fields (Fig. 9) with a total area of around 2 ha, with good results in terms of crop stand and crop yield, and strongly reduced draught requirement both in number of tillage operations and in required energy (Fig. 7). Farmers in that area also claim that in the absence of compaction by livestock hoofs and with reduced tillage, the topsoil has become particularly soft.

To enhance area-wide adoption, we agree with Ekboir (2002), who stated that although the development of no-till packages and their adoption by smallscale farmers followed different paths than for largescale farmers, the paths shared one important common feature: all successful programmes resulted from networks that worked with participatory research approaches. Wall (2007) further stresses the need for interested agents to catalyse innovation systems and, based on the activities of innovative farmers, to help overcome problems observed in farmer experiments. These highly participatory innovation systems do not develop without a catalyst: an individual or organisation with a specific interest in advancing the technology and its adoption (Wall, 2007).

# 7 Conclusions

Research on CA and CA-based resource-conserving technologies in Ethiopia generally fits into two major paradigms: (1) flat seedbed with use of herbicide and one surficial tillage operation (Ito et al., 2007); or (2) surface shaping using various (often heavy and complex) tools (Hussein et al., 2000; Abiye et al., 2002; Gebreyesus et al., 2006; McHugh et al., 2007; and many others). In both cases, indigenous in situ soil and water conservation tillage technology, if recognised, is merely mentioned anecdotally (Gebreyesus et al., 2006; McHugh et al., 2006; McHugh et al., 2006; McHugh et al., 2007), and even considered as an interference with the experiment (Hussein et al., 2000; Ito et al., 2007).

Yet, in the northern Ethiopian highlands farmers traditionally use the *marasha* and plough for in situ conservation by creating surface depressions, either at the moment of sowing (*terwah*, *derdero*) or after crop emergence (*shilshalo*). On Vertisols, the indigenous conservation technologies *terwah* and *derdero* aim to locally drain out the crops while curbing runoff and favouring infiltration.

Building upon this indigenous knowledge, we further developed CA-based resource-conserving technologies into *derdero*+ whereby the traditional ard plough was found suitable for a 'bed-and-furrow' system. The concept further integrates the retention of stubble, improving the soil structure by decreasing the number of tillage operations, whereby tillage becomes limited to transferring sediment from the furrows to the beds at the moment of sowing.

From the socio-economic point of view, implementation of permanent beds leads to decreased oxen (and hence straw) requirements, but also to increased need for weeding in the first years unless herbicide is used. These modified labour requirements need to be studied. The commonly practised stubble grazing system is incompatible with stubble management that is part of *derdero*+. It is expected that benefits from *derdero*+ will be an additional incentive for free grazing abandonment.

Furthermore, the decreased runoff (-51%) and soil loss (-81%) allow protection of the downslope areas from flooding, but soil nutrient build-up and soil structure improvement are slow processes, and hence the full benefit of the permanent bed system can only be expected after some years.

Overall, derdero+ can be part of the ongoing intensification process which includes physical soil and water conservation, slope reforestation and irrigation development, and will have to be implemented through a step-by-step process of adoption of resourceconserving technologies that lead to CA. It has its own niche: the cropped land sensu stricto, i.e. the most important part of the land, both for the farmer and for a nation that is striving for food security. Hence, it is suggested that the permanent bed *derdero+* system could be widely implemented on farmers' fields after several years of experimentation, when stable improved crop yields can be shown. The terwah system is a very good first step, resulting in decreased runoff (-11%) and soil loss (-36%), and we strongly recommend implementing it on all farmland in northern Ethiopia. However, the long-term goal is to achieve a permanent bed planting system.

In a country like Ethiopia, with an enabling environment for smallholder agriculture (Mitiku et al., 2006) and with the *marasha* as a tool that is almost ready for use, there are good prospects for CA and bed-andfurrow CA-based resource-conserving technologies.

Acknowledgements This study was carried out in the framework of Mekelle University Institutional University Cooperation, funded by VLIR, Belgium. We particularly thank the ten farmers who agreed to implement *derdero+* on their lands. Numerous other Ethiopian farmers agreed to share their knowledge with us, as well as one Australian farmer (Leigh Vial), and Solomon Gebregziabher, Tigist Oicha, Fekadu Getnet and Tewodros Gebregziabher (Mekelle University, Ethiopia), Mulugeta Gebreselassie (Bureau of Agriculture, Mekelle, Ethiopia), Wagnew Ayalneh (International Livestock Research Institute, Addis Ababa, Ethiopia), and Dirk Raes and Josse Debaerdemaeker (K.U. Leuven, Belgium). The authorities of the villages and districts concerned facilitated the research. While carrying out fieldwork, J.N. was employed by K.U. Leuven, Belgium. Soil scientist Neil Munro (Dirleton, East Lothian, UK) revised and corrected the English. The editor and two anonymous reviewers are sincerely thanked for useful comments on an earlier version of this paper.

#### References

- Abiye A., Jabbar M., Mohamed Saleem M.A., Erkossa T. (2002) Technical and economic performance of animal-drawn implements for minimum tillage: experience on Vertisols in Ethiopia, Exp. Agric. 38, 185–196.
- Asefa T., Tanner D., Bennie A.T.P. (2004) Effects of stubble management, tillage and cropping sequence on wheat production in the south-eastern highlands of Ethiopia, Soil Till. Res. 76, 69–82.
- Audebeau C., Mosséri V. (1916) Le labourage en Egypte, Bull. Inst. Egyptien 10, 88–127.
- Aune J.B., Matewos T., Fenta G., Abyie A. (2001) The ox ploughing system in Ethiopia: can it be sustained? Outlook Agric. 30, 275–280.
- Bédoucha G. (1986) Une antique tradition chez les hommes de tribu des hauts plateaux yémenites: la culture du sorgho, Techniques et Culture 8, 1–68.
- Behnstedt P. (1987) Die Dialekte der Gegend von Sa'dah (Nord-Jemen), Harrassowitz, Wiesbaden.
- Blum A. (2004) Sorghum physiology, in: Nguyen H.T., Blum A. (Eds.), Physiology and Biotechnology Integration for Plant Breeding, CRC Press, Boca Raton, FL, USA, pp. 141–224.
- Blum A., Arkin G.F., Jordan W.R. (1977) Sorghum root morphogenesis and growth. I. Effect of maturity genes, Crop Sci. 17, 149–153.
- Bradford J.M., Peterson G.A. (2000) Conservation Tillage, in: Sumner M.E. (Ed.), Handbook of soil science, CRC Press, Boca Raton, FL, USA, pp. G247–G269.
- Christoffoleti P.J., Galli A.J.B., Carvalho S.J.P., Moreira M.S., Nicolai M., Foloni L.L., Martins B.A.B., Ribeiro D.N. (2008) Glyphosate sustainability in South American cropping systems, Pest Manag. Sci. 64, 422–427.
- Deckers J., Teshome Y., Negeri A., Seyfou K., (2001) Teff, in: Raemaekers R. (Ed.), Crop Production in Tropical Africa, Directorate General for International Cooperation, Brussels, pp. 96–101.
- Descheemaeker K., Nyssen J., Rossi J., Poesen J., Mitiku H., Moeyersons J., Deckers J. (2006) Sediment deposition and pedogenesis in exclosures in the Tigray Highlands, Ethiopia, Geoderma 132, 291–314.
- Desta G., Nyssen J., Poesen J., Deckers J., Mitiku H., Govers G., Moeyersons J. (2005) Effectiveness of stone bunds in controlling soil erosion on cropland in the Tigray highlands, Northern Ethiopia, Soil Use Manage. 21, 287–297.
- Ekboir J. (2002) CIMMYT 2000–2001 world wheat overview and outlook: developing no-till packages for small-scale farmers, Centro Internacional de Mejoramiento de Maiz y Trigo (CIMMYT), México.
- Erkossa T., Stahr K., Gaiser T. (2005) Effect of different methods of land preparation on runoff, soil and nutrient losses from a Vertisol in the Ethiopian highlands, Soil Use Manage. 21, 253–259.
- Fekadu Getnet. (2006) Soil physical properties and wheat (*Triticum aestivum* L.) yield under conservation agriculture in Mayzeg-zeg watershed – Dogu'a Tembien, Unpub. M.Sc. thesis, Tropical Land Resources Management, Mekelle University, Ethiopia.
- Gebreyesus B., Wortmann C.S., Martha M., Heluf G., Amare B. (2006) Micro-Basin Tillage for Grain Sorghum Production in Semiarid Areas of Northern Ethiopia, Agron. J. 98, 124–128.

- Hussein M.A., Bjornstad A., Aastveit A.H., Berg T., Haile G. (2000) The effect of plant population and water conservation methods on grain yield of early-maturing maize varieties in the moisture-stress areas of southern Ethiopia, Afr. Crop Sci. J. 8, 159–170.
- Ito M., Matsumoto T., Quinones M.A. (2007) Conservation tillage practice in sub-Saharan Africa: the experience of Sasakawa Global 2000, Crop Prot. 26, 417–423.
- Jutzi S., Anderson F., Abiye A. (1987) Low-cost modifications of the traditional Ethiopian tine plough for land shaping and surface drainage of heavy clay soil: preliminary results from on-farm verification trials, ILCA Bull. 27, 25–31.
- Leeuwis C., Van den Ban A. (2004) Communication for Rural Innovation, Rethinking Agricultural Extension, Blackwell Science, Oxford.
- McHugh O.V., Steenhuis T.S., Berihun A., Fernandes E. (2007) Performance of in situ rainwater conservation tillage techniques on dry spell mitigation and erosion control in the drought-prone North Wello zone of the Ethiopian highlands, Soil Till. Res. 97, 19–36.
- Medina G.G., Rumayor R.A., Cabañas C.B., Luna F.M., Ruiz C.J.A., Gallegos V.C., Madero T.J., Gutiérrez S.R., Rubio D.S., Bravo L.A.G. (2003) Potencial productivo de especies agrícolas en el estado de Zacatecas, Instituto Nacional de Investigaciones Forestales, Agrícolas y Pecuarias, Centro de Investigación Regional Norte Centro, Calera, Zacatecas, México.
- Melesse T., Hoogmoed W.B., Rockström J., Savenije H.H.G. (2009) Conservation tillage implements and systems for smallholder farmers in semi-arid Ethiopia, Soil Till. Res. 104, 185–191.
- Melesse T., Rockström J., Savenije H.H.G., Hoogmoed W.B., Dawit A. (2008) Determinants of tillage frequency among smallholder farmers in two semi-arid areas in Ethiopia, Phys. Chem. Earth, Parts A/B/C 33, 183–191.
- Mitiku H., Herweg K., Stillhardt B. (2006) Sustainable Land Management – a new Approach to Soil and Water Conservation in Ethiopia, Mekelle University, Mekelle, Ethiopia, University of Berne, Berne, Switzerland.
- Muliokela S.W., Hoogmoed W.B., Stevens P., Dibbits H. (2001) Constraints and possibilities for conservation farming in Zambia, Conservation agriculture, a worldwide challenge, Proceedings 1st Wold congress on Conservation Agriculture, FAO, Madrid, pp. 61–65.
- Nyssen J., Clymans W., Poesen J., Vandecasteele I., De Baets S., Nigussie H., Naudts J., Amanuel H., Moeyersons J., Mitiku H., Deckers J. (2009) How soil conservation affects the catchment sediment budget – a comprehensive study in the north Ethiopian highlands, Earth Surf. Proc. Landf. 34, 1216–1233.
- Nyssen J., Naudts J., De Geyndt K., Mitiku H., Poesen J., Moeyersons J., Deckers J. (2008) Soils and land use in the Tigray highlands (Northern Ethiopia), Land Degrad. Dev. 19, 257–274.
- Nyssen J., Poesen J., Mitiku H., Moeyersons J., Deckers J. (2000) Tillage erosion on slopes with soil conservation structures in the Ethiopian highlands, Soil Till. Res. 57, 115–127.
- Nyssen J., Poesen J., Moeyersons J., Deckers J., Haile M., Lang A. (2004) Human impact on the environment in the Ethiopian and Eritrean highlands – a state of the art, Earth-Sci. Rev. 64, 273–320.

- Nyssen J., Poesen J., Moeyersons J., Mitiku H., Deckers J. (2008) Dynamics of soil erosion rates and controlling factors in the Northern Ethiopian Highlands – towards a sediment budget, Earth Surf. Proc. Landf. 33, 695–711.
- Nyssen J., Vandenreyken H., Poesen J., Moeyersons J., Deckers J., Mitiku H., Salles C., Govers G. (2005) Rainfall erosivity and variability in the Northern Ethiopian Highlands, J. Hydrol. 311, 172–187.
- Rajaram G., Erbach D.C., Warren D.M. (1991) The role of indigenous tillage systems in sustainable food production, Agric. Human Values 8, 149–155.
- Rezene F., Woldeyesus S., Aliye H., Asfaw N. (1992) Weed control research on maize in Ethiopia: A review, in: Benti T., Ransom J.K. (Eds.), Proceedings of the First National Maize Workshop of Ethiopia. IAR/CIMMYT, Addis Ababa, pp. 62–73.
- Rockström J., Kaumbutho P., Mwalley J., Nzabi A.W., Melesse T., Mawenya L., Barron J., Mutua J., Damgaard-Larsen S. (2009) Conservation farming strategies in East and Southern Africa: yields and rain water productivity from onfarm action research, Soil Till. Res. 103, 23–32.
- Sayre K. (2004) Raised-bed cultivation, in: Lal R. (Ed.), Encyclopedia of Soil Science, Marcel Dekker, New York.
- Simoons F.J. (1958) The agricultural implements and cutting tools of Begemder and Semyen, Ethiopia, Southwest. J. Anthropol. 14, 386–406, http://www.jstor.org/stable/3628933.
- Solomon G., Mouazen A.M., Van Brussel H., Ramon H., Nyssen J., Verplancke H., Mintesinot B., Deckers J., De Baerdemaeker J. (2006) Animal drawn tillage, the Ethiopian ard plough, maresha: a review, Soil Till. Res. 89, 129–143.
- Spaccini R., Mbagwu J.S.C., Conte P., Piccolo A. (2006) Changes of humic substances characteristics from forested to cultivated soils in Ethiopia, Geoderma 132, 9–19.
- Tadesse N., Ahmed S., Hulluka M. (1996) The effects of minimum tillage on weeds and yield of durum wheat in central Ethiopia, Trop. Agric. 73, 242–244.
- Tesfay A., Cornelis W.M., Nyssen J., Govaerts B., Tewodros G., Tigist O., Fekadu G., Raes D., Mitiku H., Deckers J., Sayre K.D. (2010) Effects of conservation agriculture on runoff, soil loss and crop yield under rain fed conditions in Tigray, Northern Ethiopia, Soil Use Manage., submitted.

- Tewodros G., Nyssen J., Govaerts B., Getnet F., Behailu M., Haile M., Deckers J. (2009) Contour furrows for in situ soil and water conservation, Tigray, Northern Ethiopia, Soil Till. Res. 103, 257–264.
- Tigist O., Cornelis W., Verplancke H., Nyssen J., Deckers J., Mintesinot B., Mitiku H. (2010) Short-term effects of conservation agriculture on Vertisols under tef (*Eragrostis tef* (Zucc.) Trotter) in the northern Ethiopian highlands, Soil Till. Res., 106, 294–302.
- Tilahun T., Ayana G., Abebe F., Wegary D. (2002) Maize pathology in Ethiopia: a review, in: Mandefro N., Tanner D., Twumasi-Afriyie S. (Eds.), Enhancing the Contribution of Maize to Food Security in Ethiopia, Proceedings of the 2nd National Maize Workshop of Ethiopia, Addis Ababa; Ethiopia, 12–16 Nov., 2001, EARO/CIMMYT, pp. 97–105.
- Vancampenhout K., Nyssen J., Desta G., Deckers J., Poesen J., Mitiku H., Moeyersons J. (2006) Stone bunds for soil conservation in the northern Ethiopian highlands: Impacts on soil fertility and crop yield, Soil Till. Res. 90, 1–15.
- Varisco D.M. (1982) The ard in highland Yemeni agriculture, Tools Till. 4, 158–172.
- Varisco D.M. (2004) Terminology for plough cultivation in Yemeni Arabic, J. Semitic Stud. 49, 71–129.
- Verhulst N., Govaerts B., Verachtert E., Castellanos-Navarrete A., Mezzalama M., Wall P.C., Chocobar A., Deckers J., Sayre K.D. (2010) Conservation agriculture, improving soil quality for sustainable production systems? in: Lal. R., Stewart B.A., Food security and soil quality, Boca Raton: CRC Press, Adv. Soil Sci., 137–208.
- Wall P. (2007) Tailoring conservation agriculture to the needs of small farmers in developing countries: an analysis of issues, J. Crop Improv. 19, 137–155.
- Wolde M., Veldkamp E., Mitiku H., Nyssen J., Muys B., Kindeya G. (2007) Effectiveness of exclosures to restore degraded soils as a result of overgrazing in Tigray, Ethiopia, J. Arid Environ. 69, 270–284.
- Woldeamlak B., Sterk G. (2003) Assessment of soil erosion in cultivated fields using a survey methodology for rills in the Chemoga watershed, Ethiopia, Agric. Ecosyst. Environ. 97, 81–93.

# Biological Nitrogen Fixation and Socioeconomic Factors for Legume Production in Sub-Saharan Africa

Jonas N. Chianu, E.M. Nkonya, F.S. Mairura, Justina N. Chianu, and F.K. Akinnifesi

#### Contents

1	Introduction	310
2	Production and Marketing of Inoculants in Sub-Saharan Africa	312
3	<b>Inoculation and Grain Legume Nodulation</b>	313
	Promiscuous Soybean Varieties	314
	and Determinants	315
	3.3 Nitrogen Fixation by Promiscuous Soybean Varieties	316
4	Success in the Use of Inoculants to Increase Yields:	
	Trials	317
5	Success in the Use of Inoculants to Increase Yields: Farmers' Fields	318
6	Limitations to the Use of Inoculants to Increase	
	Yields       2         6.1       Lessons from the Reasons for Failure in the Use of Inoculants to Increase Yields	<ul><li>319</li><li>320</li></ul>
7	<b>Constraints and Challenges to Production</b>	
	and Farmers' Use of Inoculants in SSA7.1Socioeconomic Constraints and Challenges7.2Quality Constraints and Challenges7.3Policy Constraints and Challenges7.4Other Constraints and Challenges	320 320 322 322 322 323
8	Opportunities for Production and Use of Inoculants in Africa	323
9	Highlights and Key Messages	324
10	Conclusion	325
Re	ferences	326

J.N. Chianu (🖂)

**Abstract** Low crop productivity is a general problem facing most farming systems in sub-Saharan Africa (SSA). These low yields are pronounced in grain legumes and are often associated with declining soil fertility and reduced N2-fixation due to biological and environmental factors. Unfortunately, the majority of African small farmers are now unable to afford the high mineral fertilizer prices. More than 75% of the fertilizers used in Africa are imported, putting pressure on foreign exchange. Low cost and sustainable technical solutions compatible with the socioeconomic conditions of small farmers are needed to solve soil fertility problems. Biological nitrogen fixation (BNF), a key source of N for farmers using little or no fertilizer, constitutes one of the potential solutions and plays a key role in sustainable grain legumes (e.g., soybean) production. Given the high cost of fertilizer in Africa and the limited market infrastructure for farm inputs, current research and extension efforts have been directed to integrated nutrient management, in which legumes play a crucial role. Inoculation with compatible and appropriate rhizobia may be necessary where a low population of native rhizobial strains predominates and is one of the solutions which grain legume farmers can use to optimize yields. It is critical for sustained yield in farmlands deficient in native rhizobia and where N supply limits production. Research on use of *Rhizobium* inoculants for production of grain legumes showed it is a cheaper and usually more effective agronomic practice for ensuring adequate N nutrition of legumes, compared with the application of N fertilizer. Here, we review past and on-going interventions in Rhizobium inoculation (with special reference to soybean) in the farming systems of SSA with a view to understanding the best way to effectively

TSBF-CIAT, c/o World Agroforestry Centre (ICRAF), UN Avenue, Gigiri, PO Box 30677, Nairobi, Kenya e-mail: j.chianu@cgiar.org

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_17,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Chianu et al., Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2010004

advise on future investments to enhance production and adoption of BNF and inoculant technologies in SSA. The major findings are: (1) complete absence of or very weak institutions, policy and budgetary support for biotechnology research and lack of its integration into wider agricultural and overall development objectives in SSA, (2) limited knowledge of inoculation responses of both promiscuous and specifically nodulating soybean varieties as well as the other factors that inhibit BNF, hence a weak basis for decisionmaking on biotechnology issues in SSA, (3) limited capacity and lack of sustainable investment, (4) poorly developed marketing channels and infrastructure, and limited involvement of the private sector in the distribution of inoculants, and (5) limited farmer awareness about and access to (much more than price) inoculants. The lessons learned include the need: (1) to increase investment in Rhizobium inoculation technology development, and strengthen policy and institutional support, (2) for public private partnership in the development, deployment and dissemination of BNF technologies, (3) to develop effective BNF dissemination strategies (including participatory approach) to reach farmers, (4) for greater emphasis on capacity building along the BNF value chain, and (5) for partnership between universities in SSA and those in the North on BNF research.

**Keywords** Low soil fertility  $\cdot$  N<sub>2</sub> fixation  $\cdot$  Inoculants  $\cdot$  Soybean  $\cdot$  Adoption drivers  $\cdot$  Sub-Saharan Africa

#### Abbreviations

ASK	Agricultural Society of Kenya
BCR	Benefit Cost Ratio
BNF	<b>Biological Nitrogen Fixation</b>
CAN	Calcium Ammonium Nitrate
CIAT	Centro Internacional Agricultura de
	Tropical
DAP	Di-Ammonium Phosphate
DR-Congo	Democratic Republic of Congo
FAO	Food and Agriculture Organization
	of the United Nation
FYM	Farmyard manure
IAEA	International Atomic Energy Agency
IITA	International Institute of Tropical Agri-
	culture

IFDC	International Fertilizer Development
	Center
INM	Integrated Nutrient Management
ISAR	Institut des Sciences Agronomique du
	Rwanda
KARI	Kenya Agricultural Research Institute
KIOF	Kenya Institute of Organic Farming
KShs	Kenya Shillings
LSD	Least Significant Difference
MIRCEN	Microbiological Resources Center
MRR	Marginal Rate of Returns
Ν	Nitrogen
NGO	Non-governmental organization
NPK	Nitrogen Phosphorus Potassium
OMMN	Organic Matter Management Network
Р	Phosphorus
R&D	Research and Development
SPRL	Soil Productivity Research Laboratory
SSA	Sub-Saharan Africa
SUA	Sokoine University of Agriculture
TGx	Tropical Glycine crosses
UNEP	United Nations Environment Program
UNESCO	United Nations Educational, Scientific
	and Cultural Organization
USAID	United States Agency for International
	Development
USDA	United States Department of Agriculture
US\$	United States of America Dollar

# **1** Introduction

Smallholder farming systems in sub-Saharan Africa (SSA) are constrained by low crop productivity. Grain legumes are being seen as the "meat for the poor" due to their rich protein content and the low prices of pulses compared with meat. High productivity of pulses becomes vital as most poor people in SSA depend on pulses for protein supply to meet their food, nutritional and health needs. The average yields of grain legumes have remained very low (e.g., about 622 kg/ha for common bean) under farmers' conditions compared with research station yields (e.g., over 1500 kg/ha for common beans) (Mushi, 1997). Low yields are associated with declining soil fertility due to continuous cropping without soil replenishment and reduced N2-fixation due to various biological and environmental factors (Dakora and Keya, 1997). This becomes more severe as farmers expand into marginal lands in response to population pressure. Arid and semiarid marginal lands are generally deficient in nitrogen (N), required to raise crop production (Mugabe, 1994).

Methods to enhance cost-effective soil nutrient supplies to small farmers have remained a challenge to scientists working to improve agricultural productivity in SSA. In the highland agricultural ecozones of SSA, N supply is also a key limiting factor in crop production for 35-45% of the farmers (Odame, 1997). Soybean is estimated to fix 80% of its N needs (Smaling et al., 2008). Many SSA countries have a growing need for mineral fertilizers to enhance crop yields (Mugabe, 1994; Morris et al., 2007; World Bank, 2008). However, the majority (about 60%) of African smallholder farmers are unable to afford the high prices of mineral fertilizers (Yanggen et al., 1998). A farmer has to sell about 10 kg of maize or 5 kg of common bean to buy 1 kg of N or P in the form of mineral fertilizers (Odame, 1997). This high conversion (crop to sell to purchase a unit of mineral fertilizer) ratio implies a low incentive to use fertilizers, and is largely explained by the high average farm gate prices of fertilizer in SSA. More than 75% of the mineral fertilizers used in Africa are imported (Mugabe, 1994), leading to reduced foreign exchange. For example, in the 1990s, Kenya spent over 30% of its foreign exchange annually on fertilizer importation (Mugabe, 1994).

Although grain legume yields can be improved by use of moderate levels of mineral fertilizers, legumes (unlike cereals) are rarely fertilized by farmers, probably due to the high cost of fertilizers and low awareness of the associated economic returns, or both (Ndakidemi et al., 2006). Supplementing legumes with soil nutrients has been shown to double yields (Dakora, 1984; da Silva et al., 1993), and increase plant growth and N<sub>2</sub>-fixation compared with the unfertilized control (Ndakidemi et al., 2006). Rhizobium inoculation also helps to boost the yield of grain legumes, leading to land savings, and has been described as a cheap insurance for higher yields (Ndakidemi et al., 2006). However, despite its potential to address low N and its cost effectiveness, the demand for inoculants by farmers in SSA remains low (Kannaiyan, 1993). The reasons included poor quality, and inadequate and inefficient marketing outlets, as well as inadequate extension services covering inoculant use (Kannaiyan, 1993; Odame, 1997).

Given the high cost of fertilizer in Africa and the limited market infrastructure for agro-inputs, current research and extension efforts have been directed to integrated nutrient management (INM), in which leguminous crops, shrubs and trees play a key role (Chianu and Tsujii, 2005; Mafongoya et al., 2007). BNF is a key source of N for farmers who use little or no fertilizer, especially for legumes such as soybean (Smaling et al., 2008). Rinnofner et al. (2008) described BNF, in the context of legume catch crops, as an additional benefit. Based on the studies of Zotarelli et al. (1998), Alves et al. (1999, 2005), Hungria et al. (2006) and Araujo et al. (2006), it has been widely shown that up to 80% of the above-ground N accumulation in soybean is due to N fixation by rhizobia. A study carried out in Uganda shows that BNF contributes 22% of nitrogen inflows for perennial crops and 44% for annual crops (Nkonya et al., 2008). It is estimated that about 11.1 million metric tons of nitrogen is fixed annually through BNF in developing countries (Hardarson et al., 2003). If supplied through mineral fertilizers, about twice this amount will be required to achieve the same crop yield level (Hardarson et al., 2003). Most African countries could reduce expenditures on fertilizer imports through a full exploitation of BNF (Mugabe, 1994). It is estimated that *Rhizobium* alone could provide more than 50% of the fertilizer required for crop production in most of the marginal lands of Kenya, Zimbabwe and Tanzania (Mugabe, 1994). This underscores the importance of BNF in ensuring sustainable and low cost production by smallholder farmers in SSA. However, several constraints (socioeconomic, environmental, production, etc.) need to be addressed to tap the full benefits of BNF (Bohlool et al., 1992; Amijee and Giller, 1998; Date, 1999). The most serious problems affecting nodulation and N fixation in tropical cropping systems are limited skills, low quality inoculants, acid soils and high soil temperatures, among others.

The overall BNF history, especially in SSA, indicates that holistic studies addressing the challenges facing the use of inoculants by farmers have been rare. This contributes to limited knowledge and approaches on how to stimulate widespread use of BNF. Most studies on rhizobial inoculants have tended to address the biophysical components and the science of *Rhizobium* production (Amijee and Giller, 1998). Some research has also been carried out on the economic productivity of inoculants under experimental studies (Ndakidemi et al., 2006). Only a few studies have examined the socioeconomic and policy constraints affecting household adoption and utilization of inoculants by farmers. Meanwhile, lack of knowledge by farmers is a key factor explaining the low adoption, use and retention of inoculant technologies (Date, 1999).

Here, we review past and on-going interventions in *Rhizobium* inoculation (with special reference to soybean) in SSA in order to understand how best to effectively advise on future investments to popularize soybean enterprises and enhance production and adoption of BNF and inoculant technologies to improve the farming systems, household nutrition, income and overall welfare in SSA. We hypothesize that soybean inoculation significantly enhances the sustainability and productivity of farming systems in SSA and as a result, leads to reduction in poverty, and improvements in farm income and the overall welfare of farm families.

# 2 Production and Marketing of Inoculants in Sub-Saharan Africa

BNF and N-inoculants have had a long history in Africa, starting with the colonial agricultural research efforts to develop N-inoculants for pasture legumes aimed at increasing the productivity of exotic cattle (Odame, 1997). To promote BNF in developing countries, UNESCO established a number of Microbiological Resource Centers (MIRCENs) across five continents with support from the UNEP and the FAO (Odame, 1997). The broad responsibilities of MIRCENs in Africa (located in Dakar, Cairo and Nairobi) include the collection, identification, maintenance and testing of strains as well as preparing inoculants and distributing cultures compatible with local crops. Others are the deployment of local rhizobia inoculant technology, promotion of research, and the provision of advice, training and guidance to individuals and institutions engaged in rhizobiology research. For instance, the Nairobi MIRCEN project promoted and transferred BNF technologies (including inoculants of pulses, pasture legumes and trees) to researchers and other agricultural stakeholders in Kenya and all of east Africa. The Nairobi MIRCEN also diversified into screening Rhizobium strains for

tolerance to abiotic stress (e.g., high temperatures, soil acidity, drought, etc.), especially given the numerous environmental stresses that affect the success of BNF and since two-thirds of Kenya's agricultural land suffers from these. The idea is to gradually intensify screening trials for rhizobia that are adapted to such ecological stresses. The Nairobi MIRCEN project also explored the potential of mycorrhiza, a fungal strain, on plant roots that assists the plant to extract P and water from the soil. The Nairobi MIRCEN also developed a marketable bio-fertilizer called *Biofix* (Odame, 1997). Two non-governmental organizations (NGOs) [Kenya Institute of Organic Farming (KIOF) and the Organic Matter Management Network (OMMN)] played a significant role in distributing Biofix to farmers. Over time, KIOF's active involvement in Biofix promotion over-stretched its financial and human resources and waned.

In the 1990s, the FAO supported a project to select better strains of Rhizobia in Tanzania (Mugabe, 1994). Through this, Sokoine University of Agriculture (SUA), Morogoro, developed a bio-fertilizer (Nitrosua) for use in soybean production. In collaboration with the Ministry of Agriculture and some NGOs, SUA also established extension activities to disseminate Nitrosua to local farmers. These activities also waned over time (Bala, 2008). Inoculants were produced in Uganda by at least two plants (Madhavani Ltd. and the BNF of Makerere University, established in 1990 with the help of USAID). These two plants functioned up to 1997 and, for the contract order by the FAO produced 14.2 tons of soybean inoculants between 1995 and 1997. In Rwanda, inoculant production started at the Institut des Sciences Agronomiques du Rwanda (ISAR) in 1984 and had by 1990 reached an annual production level of 2.4 tons (Cassien and Woomer, 1998). Activities were, however, disrupted by the civil war of the 1990s. Following the end of the war and the renovation of the laboratory, BNF activities resumed. However, pre-civil war levels are yet to be reached (Giller, 2001).

Commercially manufactured South African inoculants first appeared in the market in 1952 but were of doubtful quality until an independent quality control system was introduced in the early 1970s (Strijdom, 1998). Since 1976, all inoculants must be manufactured with sterilized peat and must contain at least  $5 \times 10^8$  rhizobial cells g<sup>-1</sup> of peat (Strijdom, 1998). The quality control measures ensured that South

African inoculants compared well with the best quality inoculants produced outside Africa (Strijdom and van Rensburg, 1981). A range of inoculants are produced for a number of crops including soybean, groundnut and cowpea (Deneyschen et al., 1998). In Malawi, commercially available inoculants for crops such as soybean and cowpea, produced by Chitedze Agricultural Research Station, Lilongwe, were available and sold in 50-g packets (Khonje, 1989), starting from the 1970s. Sales rose dramatically from 450 packets in 1976 to about 1800 in 1987/88. The presence of a large and well-established commercial soybean sector in Zimbabwe suggests widespread use of inoculants in the country (Mpepereki et al., 2000). The Soil Productivity Research Laboratory (SPRL) spearheaded a project in the 1990s for enhancing BNF technology in Zimbabwe, supported by the IAEA (IAEA, 1998 cited by Bala, 2008). The project developed the capacity to mass produce inoculants (120 000 packets per year) which were distributed to small farmers through the Government's extension system. The University of Zimbabwe also undertook mycorrhizal inoculation research in some regions (Mugabe, 1994).

# 3 Inoculation and Grain Legume Nodulation

Due to the importance of legumes in N<sub>2</sub>-fixation, soil fertility improvement, and human and livestock nutrition in SSA, the agricultural extension services in SSA have traditionally promoted and encouraged their inclusion in the farming systems (e.g., through promotion of cereal-legume intercropping). The cereallegume intercrop has been shown to increase the total value of cereal and legume production, land equivalent ratio and returns to labor and other inputs (Maingi et al., 2001; Dapaah et al., 2003). This suggests the suitability of the cereal-legume intercropping system for smallholder farmers with limited land and resources to buy inorganic fertilizer and other external inputs. Hence promotion of the cereal-legume intercropping system by the extension services in SSA will also increase the adoption of Rhizobium inoculation technology.

Studies have shown that *Rhizobium* inoculation is needed in all agricultural lands deficient in N and

where N supply is a key limiting factor in crop production (Odame, 1997). Examples of such lands abound in SSA (e.g., marginal lands, arid and semiarid lands, some highland areas that have lost their fertility) (Mugabe, 1994). Inoculating legumes with speciesspecific Rhizobium increases the success of their establishment, root nodulation, biomass and biomass N yields (Java et al., 1995; Zhu et al., 1998). Inoculation helps to increase the number of effective rhizobia (Boahen, 2008). It may also be necessary if a legume, newly introduced into an area, is to form effective symbiosis. Kaizzi (2002) and Giller et al. (1994) reported that Velvet bean (Mucuna pruriens) accumulated 68-220 kg N ha<sup>-1</sup> in eastern Uganda, 50% of which was derived from the atmosphere through BNF. Indigenous Bradyrhizobium spp. also does not meet the demand for N by soybean all the time. Poor yields may occur whenever effective bradyrhizobia are inadequate or even when available in adequate numbers (Ken Giller, pers. commun.). The overuse of classical N fertilizers substantially contributes to environmental degradation through groundwater pollution by nitrates (Paynel et al., 2008). This leaching of N into waters is an economic flaw since only a part of the fertilizer is used by the plants. Rhizobium inoculants have been used to address the problems of soil fertility and inadequate fertilizer application in grain legume production and have been found to be a cheaper and usually more effective agronomic practice for ensuring adequate N nutrition of legumes, compared with the application of N fertilizer (Tran, 2004; Paynel et al., 2008). Studies have also shown that under the natural BNF inhibiting conditions, inoculation is much cheaper than mineral N fertilizer (Odame, 1997). A 100-g packet of Biofix (a form of Rhizobium inoculant), sufficient to inoculate 15 kg of common bean seed, enough to plant 1 acre, costs KShs 75 (or US\$1.20). In contrast, 100 kg of inorganic CAN (Calcium Ammonium Nitrate) fertilizer, required for 1 acre, is sold for KShs 2040 (or US\$34). Apart from being cheaper (28 times) than commercially produced N fertilizers, a 100-g packet of Biofix is also lighter to transport.

Worldwide, the use of *Bradyrhizobium* in soybean has been a major success in bio-fertilizers and constitutes an example of traditional biotechnology application. However, it is important to note that unlike soybean, cowpea hardly responds to inoculation in most parts of SSA because of the presence of suitable native rhizobia in the soil.
## 3.1 Inoculation Response of Commercial vs. Promiscuous Soybean Varieties

Commercial soybean varieties are specific and require repeated inoculation due to non-persistence of the rhizobia. Corby (1965) was the first to describe the nodulation of soybean by rhizobia indigenous to African soils. He observed that a variety, 'Hernon 147', nodulated effectively with indigenous rhizobia and did not respond to inoculation in five out of six sites in Zambia and Zimbabwe. Field studies at the IITA (International Institute of Tropical Agriculture) in Nigeria showed that seed inoculation with commercial rhizobial inoculants significantly improved neither nodulation nor yield of three Asian soybean varieties, 'Malayan', 'Orba' and 'TGm 686', while the American varieties 'TGm 294-4-2371', 'Bossier' and 'Jupiter' formed very few nodules without inoculation and showed yield increases of 40 to 97% (Nangju, 1980). The Asian varieties rarely responded to inoculation even when inoculation more than doubled the yields of the American varieties (Pulver et al., 1982). Weaver and Frederick (1974) suggested that inoculant rhizobial numbers of up to 1000 cells  $g^{-1}$  soil need to be applied to soybean seed to obtain greater nodule occupancy by inoculant rhizobia than the indigenous strains.

In some cases, however, the introduced legume such as soybean could form effective symbiosis with sufficient numbers of local rhizobia, thus obviating the need for inoculation. Chowdhury (1977) reported that soybean had been grown successfully without inoculation in certain parts of Nigeria, Tanzania and DR-Congo and attributed this to the selection for and increase in efficient *Rhizobium* strains due to cultivation over the years. All the varieties tested by Chowdhury (1977) formed nodules but only those bred locally from 'Hernon' varieties or 'Malayan' (Asian variety) formed many nodules. However, Sanginga et al. (2000) noted that promiscuous soybean is incapable of nodulating effectively with indigenous rhizobia in all locations in the moist savanna zone of Nigeria. Similarly, Bala (2008) observed that it is also not clear whether promiscuous soybean cultivars are effectively nodulated by indigenous rhizobial populations in all soils and under all conditions. A study by Eaglesham (1989) showed that the promiscuous soybean varieties showed inconsistent response to inoculation. This author, therefore, concluded that it may be safer to rely on effective inoculant strains rather than breed for the ability to nodulate with indigenous rhizobial strains of unknown potential. Early studies in South Africa showed local soybean cultivars to specifically form symbiosis with B. japonicum (Van Rensburg and Strijdom, 1969). However, it is important to note that even promiscuously nodulating soybeans (that often do not require inoculation), developed and cultivated in some parts of Africa, sometimes respond to inoculation. A study carried out by Osunde et al. (2003) at five sites in the moist savanna region of Nigeria showed that promiscuous soybean varieties (Tropical Glycine cross TGx 1456-2E and TGx 1660-19F) responded to inoculation (see Table 1). However, 'Magoye', an exceptionally promiscuous line released in Zambia in 1981, nodulates readily in all soils of southern Africa where it

Promiscuity was used as a major selection criterion in the IITA for more than 10 years without in-depth

has been tested and rarely responds to inoculation in

Zambia and Zimbabwe (Mpepereki et al., 2000).

**Table 1** Nitrogen fixationand N accumulation by twopromiscuous soybeancultivars [with(out)inoculation] across five sitesin the southern Guineasavanna of Nigeria

	Stover N	Grain N	Total N <sup>a</sup>	N <sub>2</sub> fixed	N <sub>2</sub> fixed
Treatment	$(\text{kg ha}^{-1})$	$(\text{kg ha}^{-1})$	$(\text{kg ha}^{-1})$	(%)	$(kg ha^{-1})$
N source:					
Inoculated	36	131	166	50	83
Uninoculated	32	121	153	46	70
$60 \text{ kg N ha}^{-1}$	26	98	124	26	32
LSD (5%) <sup>b</sup>	4	21	25	10	11
Cultivar:					
TGx 1456-2E	30	135	165	43	70
TGx 1660-19F	33	98	131	39	50
LSD (5%) <sup>b</sup>	$NS^b$	27	31	NS	15

<sup>a</sup> Total N of above-ground dry matter; <sup>b</sup> Least Significant Difference; NS = not significant. Source: adapted from Osunde et al., 2003.

microbiological studies (Sanginga et al., 2001). Recent studies on symbiotic effectiveness of indigenous rhizobia nodulating promiscuous soybean varieties in 92 Zimbabwean soils led to the identification of three isolates with superior N<sub>2</sub>-fixing potential in the 'Magoye' variety than the commercial strain MAR 1491 (Musiyiwa et al., 2005a), although the M3 isolate was later identified as superior to the commercial strains MAR 1491 and 1495 (Zengeni and Giller, 2007). Thus, local soybean cultivars in South Africa may have had some evolutionary promiscuity, which might have been overlooked by the exclusive use of the specifically nodulating commercial varieties. Using four promiscuous soybean varieties (TGx 1485-1D, TGx 1456-2E, TGx 1448-2E and TGx 1660-19F), Okogun and Sanginga (2003) observed no significant difference between the yield of inoculated and uninoculated crops at three sites in the savanna of Nigeria even though the number of native rhizobia in soils at the three sites differed by orders of magnitude. On the basis of these results, the IITA initiated a soybean breeding program in 1978 to develop 'promiscuous' soybean varieties that nodulate with indigenous soil bradyrhizobia, thus eliminating the need for inoculation (Kueneman et al., 1984). The aim was to confer the ability to nodulate with indigenous rhizobia on the American soybean varieties, which had greater yield potential and better tolerance to diseases than the Asian varieties. This effort led to the development of over 60 TGx soybean lines. Studies aimed at establishing the degree of promiscuity of soybeans with indigenous rhizobia need to be further investigated to confirm these results (Musiyiwa et al., 2005b). According to Mpepereki et al. (2000), the symbiotic interaction between soybean genotypes and rhizobial isolates varies widely (e.g., in terms of ability to nodulate, and effectiveness in N<sub>2</sub>-fixation). All plant genotypes tested by Mpepereki et al. (2000), including varieties considered to be highly specific, nodulated with indigenous isolates in some soils.

## 3.2 Response to Inoculation: Variability and Determinants

Among others, inoculation response (e.g., soybean response to *Bradyrhizobium* inoculation) is regulated

Soybean yield increase (%)



**Fig. 1** Soybean response to inoculation as regulated by the number of indigenous rhizobia (after Thies et al., 1991)

and influenced by the number and quality of indigenous rhizobia as well as abiotic and biotic factors (Abaidoo and Woomer, 2008). Figure 1 shows that the likelihood of response to inoculation by soybean is strongly influenced by the number of effective rhizobia in the soil. Soybean growing in a soil with a small number of effective rhizobia (20–50 cells g<sup>-1</sup> soil) will likely respond to inoculation (Singleton and Tavares, 1986; Thies et al., 1991).

Weaver and Frederick (1974) suggested that for inoculation to be beneficial, nodule occupancy by inoculant strains must be more than 50%, requiring an inoculum rate of at least 1000 times the soil population. In practice, however, the presence of a large indigenous population of compatible rhizobia does not necessarily preclude response to inoculation, provided the inoculant rhizobial strains are competitive and highly effective (Giller, 2001; Osunde et al., 2003). According to Danso (1990), by increasing the numbers of inoculant strains, it is possible to increase the number of nodules they occupy.

A number of biotic and abiotic factors affect  $N_2$ -fixation (see Table 2). Symbiotic  $N_2$ -fixation is highly sensitive to water stress, a major limiting factor to legume productivity in semiarid tropics (Serraj et al., 1999). Nitrogen fixation in soybean also declines under soil moisture deficit (Devries et al., 1989;

Factor	Effect	Reference	Recommendations
High soil temperature	Reduces the survival of rhizobia in soil and inhibits nodulation and N <sub>2</sub> -fixation	Munevar and Wollum, 1982; Michiels et al., 1994	Surface mulching; placement of inoculum in deeper soil layers; select heat-tolerant strains
Soil moisture	Reduces rhizobial numbers, limits migration of rhizobia, reduces nodulation and N <sub>2</sub> -fixation	Hunt et al., 1981	Optimization of soil moisture; select moisture-stress-tolerant strains
Soil acidity	Reduces the survival of rhizobia in soil, inhibits nodulation and N <sub>2</sub> -fixation and leads to P fixation. Increases aluminum toxicity and calcium deficiency	Graham et al., 1994; Zahran, 1999; Giller, 2001	Use of acid-tolerant legume cultivars and rhizobia; liming of soil to pH at which Al and Mn are no longer toxic
P deficiency	Inhibits nodulation, N <sub>2</sub> -fixation and rhizobial growth	Gates and Muller, 1979; Cassman et al., 1981	Addition of P fertilizers, amelioration of soil acidity, inoculation with effective mycorrhiza, and selection of P-efficient cultivars
Salt stress	Reduces nodule formation, respiration and nitrogenous activity	Tu, 1981; Delgado et al., 1994	Select salt-tolerant strains
High soil N level	Inhibits root infection, nodule development & nitrogenous activity	Abdel-Wahab et al., 1996; Imsande, 1986; Arreseigor et al., 1997	Breed cultivars which are less sensitive to mineral N
Herbicides, fungicides and insecticides	Inhibits rhizobial growth; reduces nodulation and N <sub>2</sub> -fixation; deforms root hairs and inhibits plant growth	Mallik and Tesfai, 1993; Isol and Yoshida, 1990	Test the particular rhizobial inoculum and its behavior in respect of the product used before application; separate placement of rhizobia and fungicides
Competition from native organisms	Suppression of inoculation by native rhizobia	Dowlig and Broughton, 1986	Targeted research

 Table 2
 Factors limiting biological nitrogen fixation in soybean and recommendations

Kirda et al., 1989). Similarly, Sinclair et al. (1987) observed that N<sub>2</sub>-fixation in soybean was sensitive to drought, a situation that underscores the shortcomings of N<sub>2</sub>-fixation in the drylands, where 38% of the rural poor in SSA live (Ryan and Spencer, 2001). High temperature, soil acidity and salinity also inhibit growth of *Rhizobium*. Nutrient deficiencies, especially P, and the lack of efficient strains of rhizobia are also among the major factors limiting symbiotic N<sub>2</sub>-fixation (Zaman-Allah et al., 2007).

These and other factors have led to efforts to inoculate legumes. Therefore, trials for establishing the need for inoculation should include tests for the limitation of BNF by other nutrients (Bohlool et al., 1992). For example, El-Hamdaoui et al. (2003) observed that application of boron and calcium increased  $N_2$ -fixation in salt-stressed soils.

## 3.3 Nitrogen Fixation by Promiscuous Soybean Varieties

A major advantage of promiscuous soybean varieties is their indeterminate growth, resulting in the production of large biomass (Mpepereki et al., 2000). Results from measurements based on the <sup>15</sup>N isotope dilution method showed that the amount of N<sub>2</sub> fixed by five promiscuous soybean varieties planted for 2 years in the southern Guinea savanna of Nigeria averaged **Table 3** Nitrogen fixation (kg/ha) by selected breeding lines grown for two seasons in three sites with poor N in the Guinea savanna of Nigeria

	Site			
Breeding line	Gidanwaya	Mokwa	Zaria	Means
High fixers:				
1485-ID	63	71	55	64
1830-20E	67	74	42	61
1526-5E	60	64	57	60
1798-7F	54	75	43	58
1799-8F	44	72	53	57
Intermediate fixers:				
1838-10E	47	50	39	46
1831-28E	39	54	38	44
1837-6E	40	52	36	43
1833-20E	36	60	33	43
Samsoy-2	34	54	38	42
Low fixers:				
1838-5E	45	34	21	33
1805-33F	37	27	36	33
1740-3F	32	37	27	32
1837-2E	29	35	32	32
1814-2E	23	32	37	31
Means	41	55	39	45
LSD <sup>a</sup>	26	25	19	13

<sup>a</sup> LSD = least significant difference.

Source: Sanginga et al., 2000.

91 kg N ha<sup>-1</sup>, representing 46% of the total plant N. In general, however, different lines of promiscuous soybean varieties growing on the same soil vary considerably in their ability to fix nitrogen (see Table 3, further summarized in Table 4).

Differences in their efficiency for  $N_2$ -fixation among the genotypes, reproducible across sites, have also been a basis for selection in current soybean breeding work of the IITA, aimed at enhancing  $N_2$ fixation in the absence of inoculation (Sanginga et al., 2000).

## 4 Success in the Use of Inoculants to Increase Yields: Trials

Early inoculation studies indicate yield advantage in soybean grain yield in tropical Africa (e.g., Sivestre, 1970; Nangju, 1980; Bromfield and Ayanaba, 1980). Sivestre (1970) noted yields of 1440 kg ha<sup>-1</sup> in inoculated soybean compared with 240 kg  $ha^{-1}$  for the uninoculated. Bonnie (1957, cited by Bala, 2008) had reported yield increases of 80-300% with inoculation in DR-Congo. In Nigeria, a series of field experiments conducted in 1978 to screen strains of B. japonicum that were efficient in N<sub>2</sub>-fixation showed that grain yields of the American soybean cultivars Bossier and TGm 294-4 were increased by as much as 100%, while cultivars of Asian origin showed no significant response (Ranga-Rao et al., 1981). Inoculation also led to grain yield increases of 40-79% in American soybean cultivars when grown in the southern Guinea savanna of Nigeria (Nangju, 1980; Pulver et al., 1982; Ranga-Rao et al., 1984). Bromfield and Ayanaba (1980) reported that inoculation of soybean in acid sands of south-eastern Nigeria resulted in grain yield increases of 300-500% when lime was added and 270-970% in the absence of lime.

In an on-farm experiment in two districts (Moshi and Rombo) in northern Tanzania, using rhizobial inoculants (*Rhizobium tropici* strain CIAT 899 for common bean, and *Bradyrhizobium japonicum* strain USDA 110 for soybean), Ndakidemi et al. (2006) observed that at harvest, soybean and common bean growth was significantly ( $P \le 0.05$ ) greater with (brady)rhizobial inoculation compared with N and P supply or uninoculated control. Relative to uninoculated unfertilized plots, grain yields of common bean increased from inoculation alone (60–78%) and from inoculation +26 kg P/ha (82–95%); with soybean,

**Table 4**Nitrogen fixation byclass of soybean breedinglines in three sites with poorN in the Guinea savanna ofNigeria

Class of breeding line	Lower end N fixed (kg/ha)	Upper end N fixed (kg/ha)	Mean N fixed (kg/ha) across sites	Lower end breeding & site	Upper end breeding & site
Low N-fixers	21	45	32.3	1838-5E	1838-5E
				(Zaria)	(Gidanwaya)
Intermediate	33	60	43.3	1833-20E	1833-20E
N-fixers				(Zaria)	(Gidanwaya)
High N-fixers	44	75	59.6	1799-8F	1798-7F
				(Gidanwaya)	(Mokwa)

Source: derived from Sanginga et al., 2000.

there was 127-139% increase in grain yield from inoculation alone, and 207-231% from inoculation +26 kg P/ha. Thus, the combined application of bacterial inoculants and P fertilizer to soybean and common bean increased biomass production and grain yield compared with the singular use of N and P or (brady)rhizobial strains (Ndakidemi et al., 2006). Economic analysis shows that the increase in grain yield with inoculation translated into higher marginal rate of return (MRR) and profitability for soybean and common bean small farms. With common bean, relative to the control, there was 66% increase in profit with inoculation in the Moshi district and 92% in the Rombo district; with provision of supplemental P (26 kg P/ha), these profit margins rose to 84% (Moshi) and 102% (Rombo). With soybean, however, the increase in profit with inoculation was much larger, about 140% (Rombo) and 153% (Moshi). With P supplementation, these rose to 224% (Rombo) and 250% (Moshi). In Zimbabwe, studies have shown that, depending on soil fertility and rainfall, inoculant use can be more cost-effective than mineral fertilizer application (Brenner, 1996). However, Kipkoech et al. (2007, p. 18) compared the efficacy of *Rhizobium* (in groundnut cropping systems) with other soil fertility-enhancing technologies (Diammonium phosphate, DAP, NPK and Farm yard manure, FYM) and the control in western Kenya. Results show that groundnut yield under Rhizobium inoculation (1362.9 kg/ha) ranked third after yield under DAP (1800 kg/ha) and NPK (1646 kg/ha). It was, however, better than the yield under FYM (1218.5 kg/ha) and the control (1208.7 kg/ha). The benefit cost ratio (BCR) follows the same trend, with Rhizobium inoculation coming third (with a BCR of 2.5:1) after DAP (with a BCR of 3.0:1) and NPK (with a BCR of 2.8:1). With a BCR of 2.2:1, FYM even trailed behind the control (BCR of 2.4:1) (Kipkoech et al., 2007, p. 19). These results show the importance of promoting inoculant use in African agriculture, especially among resource-poor farmers who cannot afford expensive mineral fertilizers (Ndakidemi et al., 2006). Besides, economic analysis of another four trials showed that farmers gained 19% more benefits from inoculation compared with urea application (Tran, 2004). In virtually all situations, there was economic benefit of inoculation, both for the legume itself and for subsequent crops. Soybean N2-fixation has an economic value in terms of the N that it supplies to the plant from the air

J.N. Chianu et al.

which otherwise would need to come from soil and/or fertilizer sources. There is also an economic value in the residual benefits for soil N fertility and increased productivity of subsequent crops (Tran, 2004). In most soils, the savings were equivalent to 40–60 kg N/ha. In a report by Duong et al. (1984), 240 kg N/ha would have been required to produce an equivalent grain yield to the inoculated treatment.

The key lesson learned with the successes of *Rhizobium* inoculation based on on-station experiments or experiments simulating farmers' conditions is that it does not really guarantee widespread uptake by the smallholder farmers, no matter how attractive the results are.

## 5 Success in the Use of Inoculants to Increase Yields: Farmers' Fields

Based on the obvious benefits of inoculation of legumes with suitable Rhizobium strains, a lot of trial research has been directed at BNF in SSA, with significant advances made. This section reviews the farm-level adoption of inoculants and the growing of promiscuous soybean varieties in Africa. Smallholder farmers in many parts of the moist savanna of Nigeria have widely adopted promiscuous soybean varieties, especially the high-yielding TGx 1448-2E (Manyong et al., 1998; Sanginga et al., 2001). In southern Africa, the exceptionally promiscuous nature of 'Magoye' led to its widespread promotion and adoption (Mpepereki et al., 2000). However, inoculating soybean with effective rhizobia is a key strategic research intervention that has contributed to significant improvement in the productivity of soybean in SSA. The development of local production of inoculants in Rwanda led to rapid expansion of the area under soybean production (Saint Macary et al., 1986). In Zimbabwe, a soybean promotion program including the use of inoculants saw to its widespread adoption by smallholder farmers, with farmer education on inoculation being a major driving force for the success of the program (Marufu et al., 1995). Presently, over 55 000 smallholder farmers in Zimbabwe inoculate and grow soybean as a result of this promotion (Mpepereki, pers. commun.). Initially designed to promote the cultivation of 'Magoye', farmers' improved access to seeds and

inoculants encouraged them to adopt both 'Magoye' and specifically nodulating cultivars. Farmers were keen to plant the latter for their high yields and income generation, and 'Magoye' for its non-requirement for inoculants as well as its large biomass used as fodder and for soil fertility maintenance and improvement. Overall, the trial discoveries have led to very few opportunities for enhancing grain legume (e.g., soybean) production among smallholder farmers in Africa due to limited farm-level availability and adoption of inoculants (Bala, 2008). There is, therefore, relatively little evidence (across SSA) to show any substantial inoculation practice among smallholder farmers.

Key lessons learned on the drivers of success at farmers' field level indicate that these include: (i) widespread demonstration of the inoculants with particular attention to the needs of small farmers; (ii) within-country collaboration and involvement of media agencies; (iii) well-coordinated collaborative research-for-development efforts; (iv) involvement of top government officers and decision-makers; (v) joint efforts of a number of national, international and private-sector organizations over many years; (vi) involvement of the private sector in production and dissemination of the inoculants, and (vii) farmer education on inoculation. These strategies have worked well in pilot areas and should be scaled up and out to reach more smallholder farmers that produce the bulk of the food (especially grain legumes) eaten in SSA. This scaling could be attained through an appropriate innovation platform involving all stakeholders. Appropriate incentives are also needed to support the private sector and industries for an effective role and delivery in the scaling up of the inoculant technology.

# 6 Limitations to the Use of Inoculants to Increase Yields

There have been several failures in the effort to use inoculants to increase the yields of grain legumes in different parts of the world, especially SSA. Adoption and use of bio-fertilizers is very limited among small farmers in many African countries such as Kenya (Odame, 1997). This author also noted that many of the modern biotechnology products (Bt. Maize, transgenic sweet potato, genetically-engineered livestock vaccines, etc.) have yet to be used by farmers. Private technology markets are undeveloped in many SSA countries (Brenner, 1996). Therefore, specific measures, such as tax incentives and exemptions, will be needed to stimulate the development of BNF technology markets and the creation of local firms (Brenner, 1996). Kueneman et al. (1984) described inoculation of soybean with specific strains of Bradyrhizobium japonicum as an investment that most farmers cannot afford. With very few exceptions, countries in SSA do not have industries to produce viable inoculants at prices that smallholder farmers can afford (Kueneman et al., 1984). In West Africa, other than experiments that are limited to research farms, there is hardly any country where rhizobial inoculation is commonly undertaken by farmers. The Dakar (Senegal) MIRCEN, which has a mandate for inoculant production, does not appear to be actively involved in either inoculation trials or the production of inoculants. The most visible BNF endeavors appear to come from East and Southern Africa. In Zimbabwe, the cultivation of specifically nodulating soybean with inoculants was promoted among smallholder farmers in the 1980s. However, when the project ended, soybean cultivation failed to make any appreciable increase, largely because of farmers' difficulty in accessing seed and inoculants (Mpepereki et al., 2000). In Kenya, in line with the obvious potential of *Biofix* to replace the often unavailable and expensive mineral fertilizers, together with the Agricultural Society of Kenya (ASK), the Nairobi (Kenya) MIRCEN sells inoculants to farmers. Between 1992 and 1993, average sales were estimated at 1350 kg per year (Mugabe, 1994). However, cutbacks in government expenditure as a part of the structural adjustment policies resulted in a 40% reduction in financial support to extension programs, affecting their agricultural programs (Mugabe, 1994). This demonstrates the unsustainable dependence on government as a major consumer of the inoculants and the need to commercialize inoculant marketing and consumption. There is also the lack of specific policy incentives to stimulate the involvement of the private sector at all stages of the innovation process to induce adoption (Brenner, 1996).

Experiences from Zambia show that the technical feasibility of *Rhizobium* inoculants for common beans remains doubtful (Sakala, 1990; Mvula et al., 1996). Elsewhere, it has been noted that, except for soybean, responses to inoculation are sporadic (Silver, 1976), mainly due to the presence of an adequate and aggressive native rhizobial population (Woomer et al., 1997), ineffective strains (Parker, 1977) or competition from indigenous rhizobial flora (Silver, 1976; Miller and May, 1991; Woomer et al., 1997). In Malawi, inoculant use has not been widely adopted by smallholder farmers, largely because they are mostly not well informed of the technology. Another problem is the fact that inoculants require refrigeration, an investment that is obviously beyond an average smallholder farmer (Khonje, 1989).

## 6.1 Lessons from the Reasons for Failure in the Use of Inoculants to Increase Yields

Some key lessons have been learned on the causes of failures in the use of inoculants to increase the yield of grain legumes. These include: (i) a disconnect between farmers' problems and biotechnological emphasis, (ii) the path from research to development and to dissemination of a biotechnology product is fraught with uncertainty (Brenner, 1996), (iii) policy and institutional arrangements are critical for widespread dissemination and adoption of Rhizobium inoculants, (iv) increase in the involvement of other local actors (e.g., private entrepreneurs, NGOs, etc.) may be more sustainable interventions for BNF distribution than total reliance on government agencies (Odame, 1997), (v) the problems of Biofix were caused by a variety of technical factors that include: complexity of tropical soils, selectivity of Rhizobium strains in infecting specific legume species, residual effects of inoculation, and the need for phosphate fertilizers to stimulate nodulation and nitrogen fixation. The other lessons are: (vi) farmers' concerns on the efficacy of Rhizobium inoculants (e.g., Biofix) in increasing crop yields as well as their operational feasibility, (vii) inadequate participatory and interactive approach among key stakeholders (including farmers) in the development of Rhizobium inoculation technology, (viii) production and marketing of inoculants (e.g., Biofix) was centralized, affecting effective distribution, (ix) weak linkages with private-sector manufacturers, local stockists, NGOs and farmers, and (x) inadequate capacity building and technical assistance for African Universities in Rhizobium inoculation technology development.

## 7 Constraints and Challenges to Production and Farmers' Use of Inoculants in SSA

This section reviews the social, economic, cultural and policy constraints and challenges to the production and farmers' use of Rhizobium inoculants in SSA. These, including their effects and possible recommendations, are summarized in Table 5. Due to limited capacity many national agricultural research systems in SSA lack the skills to set priorities in the application of biotechnology. This situation hampers the development of BNF technologies. Current research and development programs in Africa are often isolated, with little coordination, coupled with inadequate funding. Most of the time, these organizations are not needdriven and lack the ability to develop specific products. Besides, there is weak protection of intellectual property rights in many parts of Africa, hampering innovations, inventions, investments, and development of new technologies including biotechnologies.

## 7.1 Socioeconomic Constraints and Challenges

A successful transfer of *Rhizobium* inoculation technology from the laboratory to farmers' field depends on some crucial interactions among many players (research, policy, etc.). Farmers' access to inoculants remains the most controversial phase in the evolution of *Rhizobium* inoculation technology (Woomer et al., 1997) and explains why despite its acclaimed attributes the use of inoculants among smallholder farmers has been limited, making it a technology with a low rate of adoption.

Most farmers in SSA are not aware of inoculants or that legumes fix N in their nodules, yet traditional and modern farming systems almost invariably include grain legumes. Contrary to the thinking of Kueneman et al. (1984), the cost of inoculants is probably not usually a constraint to farmers who normally set aside some funds for seeds that are clearly more costly than *Rhizobium* inoculants. However, for farmers that use non-commercial seeds, there may be little incentive to purchase inoculants. Poshiwa et al. (2006) found that inoculant awareness in Zimbabwe was extremely

Socioeconomic and	Effect	Reference	Recommendations
Limited farmer awareness of and access to inoculants	Low adoption and use of inoculants in farming systems	Odame, 1997; Woomer et al., 1997; Kipkoech et al., 2007	Private sector involvement
Poor quality control of inoculants	Low viability of <i>Rhizobium</i> inoculants and uncertain performance	Johnson et al., 1994; Odame, 1997	Cold storage, use of modern technologies, and more research
Lack of trained personnel	Limited awareness by farmers of the existence of BNF (including inoculants)	Kannaiyan, 1993; Odame, 1997; Woomer et al., 1997	Raise farmer awareness about legume root nodules; familiarize farmers with <i>Rhizobium</i> inoculants
Fear over possible human and livestock health risks of inoculants by farmers	Limited adoption and use of <i>Rhizobium</i> inoculants to increase legume productivity	Odame, 1997	Involvement of farmers in the process of development of inoculants; participatory approach
Absence of policy or weak policy support and insufficient biotechnological framework	Forestalls widespread adoption; weak development of the production and marketing of inoculants	Mugabe, 1994; Odame, 1997; Silver and Nkwiine, 2007	Include the issue of bio-fertilizers in governments' effort towards addressing the problems of low and declining soil fertility
Limited scientific expertise, applied BNF brain drain, and poor research funding	Limited production of inoculants and low quality inoculants	Brenner, 1996; Ndakidemi et al., 2006	Linkages between Universities in SSA with those in the North with expertise in <i>Rhizobium</i> science; government policy support

Table 5 Socioeconomic and policy constraints to the use of *Rhizobium* inoculation technology in sub-Saharan Africa and possible intervention measures

low (2%). In central Kenya, among farmers who cultivated grain legumes, less than 15% used inoculants (Woomer et al., 1999). Although Poshiwa et al. (2006) and Woomer et al. (1999) collected data on social characteristics (e.g., age, gender, education) of the heads of the households surveyed, none of them linked these characteristics with the adoption of BNF, critical information required to explain better the low use being widely observed. The BNF adoption parameters examined by Woomer et al. (1999) includes: legume cultivation, nodule awareness, inoculation benefit awareness and inoculant use. There is the need to expand the knowledge base on BNF utilization among farmers in SSA beyond binary measures (e.g., awareness or use) to include more qualitative aspects of farmers' knowledge, willingness to pay and the longterm relevance of inoculants in farm objectives. Generally, it is necessary to understand the structure and function of knowledge held by farmers, especially the adopters of inoculants, so as to bridge the gaps between farmers' and scientific knowledge.

Brain drain, lack of trained personnel and inadequate resources have also been noted to be a major concern (Odame, 1997), just like the lack of capacity by scientists to understand the full potential for BNF application in local agro-ecosystems. The capacity building efforts on BNF have also been challenged by the mass exodus of plant biologists from applied BNF to other fields (Bohlool et al., 1992). It is clear that lack of complete understanding of legume BNF interactions under diverse farming systems contributes to the low understanding among experts in particular settings. Besides, a vast majority of smallholder farmers in SSA are not aware of the existence of BNF (including inoculants) due to lack of resources to create awareness in BNF and improve its distribution (Odame, 1997). This explains why researchers at the Nairobi MIRCEN project attributed the limited use of Biofix to "technological ignorance of smallholders", poor communication, and limited understanding of Rhizobium and plant biology (Kannaiyan, 1993; Woomer et al., 1997). They emphasized the need to raise farmer awareness

about legume root nodules (in which *Rhizobium* bacteria live) as part of the familiarization with *Rhizobium* inoculants (Woomer et al., 1997).

There are problems related to inoculant packaging size, influencing uptake potential by farmers. The minimum quantity of inoculants that a farmer has to buy is a key issue. A 100-g packet of inoculants (e.g., Biofix) is needed for 15–16 kg of common bean seed (for instance), sufficient for planting one acre of land. However, few smallholders can plant such a relatively large area with common bean. On average, farmers require only 25 g of Biofix - sufficient for 2-3 kg of common bean seed, to be planted on 0.25 acres of land, which is the approximate farm size usually allocated to legumes in SSA. Thus, if the packet size is not reduced, each packet needs to be opened and shared among several farmers, raising the issue of high administrative and other costs. Lastly, there were fears over possible health risks to young children and domestic animals. Participants at a focus group discussion described Biofix as 'poison', the leftover of which they were unsure of what to do with (Odame, 1997).

#### 7.2 Quality Constraints and Challenges

The uncertain performance of Rhizobium inoculants may explain the limited farmer adoption despite their potential to reduce the mineral fertilizer requirement, and reduce cost of production, and the fact they are lighter to transport and are more environmentally friendly than mineral N fertilizers (Odame, 1997; Kipkoech et al., 2007). Poor quality control in inoculant production processes as well as transportation and storage problems negatively affect the viability of inoculants (Odame, 1997). Based on a report for the World Bank, difficulty in obtaining and keeping inoculants was one of the factors that constrained soybean production in Zimbabwe (Johnson et al., 1994). Cold storage which might improve its viability is beyond the reach of smallholder farmers. Another constraint has to do with the limited production of inoculants due partly to limited scientific expertise, inadequately funded agricultural research and extension, and poor infrastructure (Ndakidemi et al., 2006). Limited investment of public funds in R&D and the diffusion of new technology through the national extension systems add to this problem (Brenner, 1996).

#### 7.3 Policy Constraints and Challenges

Rhizobium inoculation has the potential to offer more environmentally friendly agricultural production than the mineral fertilizer-intensive model (Brenner, 1996). However, this will only be realized if certain conditions that involve difficult policy choices and tradeoffs are met (Brenner, 1996). From a policy point of view, the constraints and challenges range from a complete absence of to very weak policy support for BNF technologies. Research into the basic mechanisms of BNF processes is an important goal for improving N<sub>2</sub>fixation. Much knowledge on BNF has been gathered with several successes documented, especially in developed countries. However, bio-technology research in SSA has generally been science-driven, ad hoc (and non-holistic), and not integrated into the wider agricultural and overall development objectives (Brenner, 1996). It is concentrated in universities which have little tradition of interaction with farmers and the private sector. The focus of scientists has been on the supply side of biotechnology (Brenner, 1996). Even in that area, there are still issues widely unattended to, including bio-safety and intellectual property rights.

Lack of policy and political support at the national level has been blamed for forestalling widespread adoption of *Rhizobium* inoculation and other suitable forms of biotechnology (e.g., germplasm) in SSA (Odame, 1997; Silver and Nkwiine, 2007). Efforts of many governments to address soil fertility problems in SSA have mostly been directed to mineral fertilizer subsidies (Jayne and Boughton, 2006). The limited commitment to non-fertilizer soil fertility management options has contributed to the weak development of the production and marketing of inoculants in SSA. For instance, *Nitrosua*, developed by SUA, could not be effectively disseminated to local farmers in Tanzania due to poor government budgets for agricultural extension (Mugabe, 1994).

In many SSA countries, bio-fertilizers are not fully addressed in national fertilizer recommendations (Odame, 1997). Only a few SSA countries (e.g., Zimbabwe) have set up national biotechnology institutions and implemented bio-safety procedures (Brenner, 1996). Besides, the question of IPRs (Intellectual Property Rights) related to biotechnology is still unresolved in most countries (Brenner, 1996).

The current research agenda on BNF has evolved due mostly to individual efforts with limited

government policy support (Mugabe, 1994). African governments must reconsider their policy on agricultural biotechnology research so as to provide a strong institutional basis for BNF (Mugabe, 1994). In the same vein, although genetically modified plants could result in increased productivity, serious concerns (e.g., health and environmental safety, etc.) have been raised over their R&D policy process in SSA (Odame, 1997). To efficiently develop, produce and disseminate new technologies, an elaborate value chain [to enhance scaling up and involving key stakeholders (researchers, farmers, extension agents, agro-input dealers, etc.)] adequately backed up with relevant policies is required. There is limited linkage among inoculant stakeholders presently. Although such linkages are poor in most other agricultural technologies in SSA, the inoculants' value chain is probably among the most poorly developed value chains, limiting its trade and dissemination.

#### 7.4 Other Constraints and Challenges

These do not strictly fall into either socioeconomic, quality or policy categories. The most important has to do with decision-making about the level of biotechnology required in different SSA countries. These challenges fall into: (i) development of the knowledge base for appropriate decision-making on use of biotechnology approaches, (ii) lack of sufficient studies that have clearly analyzed biotechnology-related data to assist policy-makers on Rhizobium-related policies, (iii) lack of priority setting for biotechnology aimed at solving problems of national importance, (iv) capacity development, and (v) the establishment of linkage and cooperative mechanisms for biotechnology development, transfer, and sustainable applications. The emphasis given to Rhizobium in bio-fertilizer research shows its high specificity to only legumes (common bean 47%; lucerne 23%; soybean 14%; desmodium (a leguminous pasture species) 9%; and other minor legumes 7%; in that order) unlike mycorrhiza, that works in 80% of all plants (Odame, 1997). A wider use of Rhizobium inoculants in marginal areas depends on the ability to develop strains which are tolerant to high temperatures, soil acidity, drought and salinity (Odame, 1997).

## 8 Opportunities for Production and Use of Inoculants in Africa

Given that widespread decline in soil fertility and agricultural productivity has largely been blamed for widespread poverty in Africa and that biotechnology has great potential to lead to increases in agricultural production, as has been demonstrated in some of the instances in this review article, we see a great opportunity for increases in the production and use of Rhizobium inoculants in the farming systems of SSA. The recent (2008) increases in the price of mineral fertilizers (even over and above the oil prices), driven by growing demand for mineral fertilizers resulting from commercial responses to increasing food prices (IFDC, 2008; Nehring et al., 2008) have also reinforced the need to develop alternative soil fertility management strategies. The price of nitrogen-based fertilizers rose from US\$277 per ton in January 2007 to over US\$450 per ton in August 2008 (IFDC, 2008). The dramatic increase and soaring prices of food reveal the large potential for developing and disseminating BNF and inoculation technologies to poor farmers who cannot afford the high fertilizer prices. The increasing concern about environmental pollution also presents an opportunity for increasing efforts to develop and promote BNF and inoculation. It is estimated that only 30-40% of fertilizer applied worldwide is used by plants. The remainder is lost, especially through leaching and volatilization, etc. (Hardarson et al., 2003).

However, these opportunities can be exploited if SSA countries develop long-term policies on BNF and biotechnology in general. Such policies should: (i) promote national biotechnology need assessment and implementation, (ii) target biotechnology research and execution to needs, (iii) provide incentives and environment for commercialization of biotechnology research and enterprises, (iv) promote partnerships among local public R&D and foreign industries in biotechnology, (v) improve scientific capacities and technological infrastructure for optimal biotechnology execution, and (vi) integrate biotechnology risk management into existing environmental, health and agricultural regimes. Otherwise, the potential benefits of biotechnology (e.g., Rhizobium inoculation) may not be tapped for the improvement of human welfare in SSA. Besides, policy related to biotechnology (e.g., BNF, Rhizobium inoculation) should address

the need to: (i) strengthen institutions (e.g., agricultural extension, NGOs) that serve the interests of smallholder farmers as they adopt biotechnology, (ii) enhance their capacity, and (iii) improve their participation in adapting and testing BNF and *Rhizobium* inoculation technologies. Instead of ad hoc approaches, SSA countries must opt for an integrated approach for biotechnology, which also requires policy intervention (Brenner, 1996). The integrated approach will ensure that biotechnology research is at the service of problems confronting smallholder agriculture.

#### 9 Highlights and Key Messages

Given the unsettled state of knowledge on the response of different varieties of soybean to inoculation and the fragmentary evidence of its economic benefits at the farm level, there is the need to initiate further studies on inoculation response of promiscuous soybean varieties as well as the commercial or specifically nodulating soybean varieties. Holistic studies addressing the challenges facing the use of inoculants by farmers have been rare in SSA. For instance, unlike experimental data, only a few studies have examined the socioeconomic and policy constraints affecting household adoption and utilization of inoculants by farmers. Trials for establishing the need for inoculation should include tests for the limitation of BNF by other nutrients (e.g., boron, calcium, etc.).

Secondly, there is also a pressing need for more indepth analysis of short- and long-term economic and social costs and benefits of *Rhizobium* inoculation and the need to expand the knowledge base on BNF utilization among farmers in SSA beyond binary measures (e.g., awareness or use) to include more qualitative aspects of farmers' knowledge, willingness to pay and the long-term relevance of inoculants in farm objectives.

Thirdly, policies and institutions promoting the development of soybean inoculants and widespread farmer adoption for increased production of both promiscuous and commercial soybean varieties are needed. This must be accompanied by targeted research to effectively address the specific needs of specific soybean growing areas in SSA. This is because, despite its potential to address low N and its "assumed" cost effectiveness and importance in ensuring sustainable and low cost production by smallholder farmers in SSA, the demand for inoculants remains low for reasons that include poor quality, and inadequate and inefficient marketing channels and outlets, as well as inadequate extension services covering inoculant use. In virtually all situations, there was economic benefit of inoculation, both for the legume itself and for subsequent crops. The lesson learned with the successes of Rhizobium inoculation based on on-station experiments or experiments simulating farmers' conditions is that it does not really guarantee widespread uptake by the smallholder farmers, no matter how attractive the results are. Like Rhizobium inoculants, many of the modern biotechnology products (Bt. Maize, transgenic sweet potato, geneticallyengineered livestock vaccines, etc.) have yet to be used by SSA farmers, especially given that private technology markets are still highly undeveloped in many SSA countries. Specific measures, such as tax incentives and exemptions, will be needed to stimulate the development of BNF technology markets and the creation of local firms. There is the need for specific policy incentives to stimulate the involvement of the private sector at all stages of the innovation process to induce adoption. The trial discoveries have led to very few opportunities for enhancing grain legumes (e.g., soybean) production among smallholder farmers in Africa due to limited farm-level availability and adoption of inoculants (Bala, 2008). There is, therefore, relatively little evidence (across SSA) to show any substantial inoculation practice among smallholder farmers. There is the need for SSA countries to develop long-term policies on BNF and biotechnology in general.

Fourthly, the review article indicates that Rhizobium inoculation is needed in all agricultural lands deficient in N and where N supply is a key limiting factor in crop production, especially since it is not clear whether promiscuous soybean cultivars are effectively nodulated by indigenous rhizobial populations in all soils and under all conditions (e.g., soil fertility and rainfall status). Although smallholder farmers in selected countries in SSA (e.g., Nigeria, Zimbabwe, Zambia, etc.) have widely adopted promiscuous soybean varieties (e.g., TGx 1448-2E, Magoye, etc.), inoculating soybean with effective rhizobia is still a key strategic research intervention that has the potential to significantly contribute to noticeable improvements in the productivity of soybean in SSA. However, some exceptionally promiscuous soybean varieties (e.g., Magoye) nodulate readily in most soils and rarely respond to inoculation. Results also confirmed the importance of promoting inoculant use in African agriculture, especially among resource-poor farmers who cannot afford expensive mineral fertilizers (e.g., DAP, NPK, etc.).

Fifthly, experiences from some SSA countries (e.g., Zambia) show that the technical feasibility of Rhizobium inoculants for some grain legumes (e.g., common beans) remains doubtful. It has also been noted that, except for soybean, responses to inoculation are sporadic, mainly due to the presence of adequate and aggressive native rhizobial population, ineffective strains or competition from indigenous rhizobial flora. Finally, there is the need to involve other local actors (e.g., NGOs, agro-input dealers, etc.) to ensure more sustainable interventions for BNF distribution than total reliance on government agencies. There is also widespread lack of skills to set priorities in the application of biotechnology in SSA, hampering the development of BNF technologies and often leading only to isolated cases with little coordination and inadequate funding. The increasing concern about environmental pollution also presents an opportunity for increasing efforts to develop and promote BNF and inoculation.

#### **10 Conclusion**

Several conclusions could be drawn from the outcomes of this review article. Firstly, while in principle, privatization of public sector institutions and activities is being encouraged, biotechnology R&D in SSA is presently focused on improving agriculture with over 85-90% of the biotechnology R&D in the region still within the public sector. Little or no incentives have been provided to encourage private sector interest in the development and application of biotechnology products. Except for South Africa, development, production and dissemination of biotechnology are limited to public institutions. Secondly, most issues around Rhizobium inoculation in parts of SSA still revolve around research with little or no effort to disseminate the results and products. Virtually no attention has been given to the demand side and to the ways in which biotechnology could best contribute to solving farmers' production problems. Most research activities and information generated started and ended with scientists who are yet to finalize even on issues of quality control. There were unsustainable interventions in the Rhizobium inoculant distribution, due to too much reliance on government agencies and little or no serious involvement of local actors (e.g., private entrepreneurs, non-governmental organizations, farmers' cooperatives, local stockists, trade associations, etc.). A coordinated program is needed to develop a capacity to produce and supply sufficient quantities of high quality inoculants to farmers and to educate extension workers and farmers about the benefits of inoculation. Thirdly, although bio-fertilizers could potentially benefit smallholder farming systems in SSA, policy support necessary to create a suitable environment for the active participation of non-government agency players is critical to ensure widespread production, distribution and application of bio-fertilizers. Unfortunately, such policies are nearly absent in most SSA countries. Fourthly, bio-fertilizer research in SSA has been science-driven, ad hoc, uncoordinated and unintegrated. Attention has essentially been focused on the supply side (e.g., training of scientists in molecular biology, biochemistry, microbiology, etc.). Little effort has been made to set clear priorities and to integrate research efforts with the broader objectives set for agricultural research. There is little interaction among the multiple actors and institutions (e.g., biotechnologists and plant breeders; the public and private sectors; scientists and farmers; and among the scientific community and policy-makers) most closely involved with biotechnology R&D and diffusion. Indeed, public research institutions at times tend to compete (rather than collaborate) for scarce resources. More importantly, there is little interaction and/or collaboration between the public and private sectors. Fifthly, efforts to develop rhizobial inoculants also need to be accompanied by research that facilitates their efficient use that is relevant to resource-poor farmers. Breeding programs can also help to develop crop varieties that nodulate under abiotic stresses. Finally, there is a strong need to train policy-makers and scientists to increase their awareness of the need to reshape biotechnology towards the needs of smallholders, and for farmer participation in prioritizing areas to be addressed by biotechnology research and policy.

Acknowledgements The authors wish to thank Dr. Abdullahi Bala of the Federal University of Technology, Minna, Nigeria and Professor Ken Giller of Wageningen University for their critical review of and constructive comments on the earlier versions of this review article.

#### References

- Abaidoo R., Woomer P. (2008) Optimizing nitrogen fixation in soybean genotypes in Africa: the relevance of biophysical constraints, PowerPoint presentation during an International Workshop on *Rhizobium* Inoculation, held at Impala Hotel, Arusha Tanzania, 17–21 March 2008.
- Abdel-Wahab H.H., Zahran H.H., Abd-Alla M.H. (1996) Root hair infection and nodulation of four legumes as affected by the form and the application time of nitrogen fertilizer, Folia Microbiol. 41, 303–308.
- Alves B.J.R., Lara-Cabezas W.A.R., David E.A., Urquiaga S. (1999) Balanco de N em soja estabelecida em um Latossolo Vermelho Escuro do triangulo mineiro em condicoes de plantio direto e prepare convencional do solo, in: Congresso Brasileiro de Ciencia do Solo, Brasilia, Brazil.
- Alves B.J.R., Zotarelli L., Araujo E.S., Fernandes F.M., Heckler J.C., Medeiros A.F.A., Boddey R.M., Urquiaga S. (2005) Balanco de N em rotacao de culturas sob plantio direto em Dourados, Boletim de Prequita e Desenvolvimento 07, Embrapa Agrobiologia, Seropedica, Brazil.
- Amijee F., Giller K.E. (1998) Environmental constraints to nodulation and Nitrogen fixation of *Phaseolus vulgaris* L. in Tanzania. I. A survey of soil fertility, root nodulation and multi-locational responses to *Rhizobium* inoculation, Afr. Crop Sci. J. 6, 159–169.
- Araujo E.S., Jantalia C.P., Boddey R.M., Urquiaga S., Alves B.J.R. (2006) Importancia do N das raizes da soja para a produtividade da cultura suceeora e para o balance de N do sistema Cicular Tecnica 14. Embrapa Agrobiologia, Seropedica, Brazil.
- Arreseigor C., Minchin F.R., Gordon A.J., Nath A.K. (1997) Possible cause of the physiological decline in soybean nitrogen fixation in response to nitrate, J. Exp. Bot. 48, 905–913.
- Bala A. (2008) Recent advances in soybean inoculum research and applications: Towards enhancing productivity in smallholder agriculture, Paper presented at an International Workshop on *Rhizobium* Inoculation, held at Impala Hotel, Arusha Tanzania, 17–21 March 2008.
- Boahen S.K. (2008) Inoculation response: Overcoming the challenges posed by indigenous rhizobia, PowerPoint presentation during an International Workshop on *Rhizobium* Inoculation, held at Impala Hotel, Arusha Tanzania, 17–21 March 2008.
- Bohlool B., Ladha J., Garrity D., George T. (1992) Biological nitrogen fixation for sustainable agriculture: A perspective, Plant Soil 141, 1–11.
- Brenner C. (1996) Policy options for developing countries: Getting into the agricultural biotechnology act? Biotechnology and Development Monitor No. 29, December.
- Bromfield E.S.P., Ayanaba A. (1980) The efficacy of soybean inoculation on acid soil in tropical Africa, Plant Soil 54, 95–106.
- Cassien B., Woomer P.L. (1998) Recent history of the BNF activities at the Institut des Sciences Agronomique du Rwanda, in: Dakora F.D. (Ed.), Biological Nitrogen Fixation in Africa: Linking Process to Progress, African Association for Biological Nitrogen Fixation, Cape Town, South Africa, pp. 33–34.

- Cassman K.G., Munns D.N., Beck D.P. (1981) Phosphorus nutrition of *Rhizobium japonicum*: strain differences in storage and utilization, Soil Sci. Soc. Am. J. 45, 517–523.
- Chianu J., Tsujii H. (2005) Integrated nutrient management in the farming systems of the savannas of northern Nigeria: what future? Outlook Agric. 34, 197–202.
- Chowdhury M.S. (1977) Response of soybean to *Rhizobium* inoculation at Morogoro, Tanzania, in: Ayanaba A., Dart P.J. (Eds.), Biological Nitrogen Fixation in Farming Systems of the Humid Tropics, John Wiley & Sons, London, pp. 245–253.
- Corby H.D.L. (1965) Report of national *Rhizobium* trial for soyabeans, 1963/64, Mimeo 236/59, Grasslands Research Station, Marandellas, Zimbabwe.
- Dakora F.D. (1984) Nodulation and nitrogen fixation by groundnut in amended and un-amended field soils in Ghana, in: Ssali H., Keya S.O. (Eds.), Biological nitrogen fixation in Africa, pp. 324–339.
- Dakora F.D., Keya S.O. (1997) Contribution of legume nitrogen fixation to sustainable agriculture in sub-Saharan Africa, Soil Biol. Biochem. 29, 809–817.
- Danso S.K.A. (1990) Evaluation of biological nitrogen fixation in plants, in: Gueye M., Mulongoy K (Eds.), Maximiser la Fixation Biologique de L'Azote pour la Production Agricole et Forestière en Afrique, pp. 258–271.
- Dapaah H., Asafu-Agyei J., Ennin S., Yamoah C. (2003) Yield stability of cassava, maize, soybean and cowpea intercrops, J. Agric. Sci. 140, 73–82.
- da Silva P.M., Tsai S.M., Bonetti R. (1993) Response to inoculation and N fertilization for increased yield and biological nitrogen fixation of common bean (*Phaseolus vulgaris* L.), Plant Soil 152, 123–130.
- Date R.A. (1999) Inoculated legumes in cropping systems of the tropics, Field Crops Res. 65, 123–136.
- Delgado M.J., Ligero F., Lluch C. (1994) Effect of salt stress on growth and nitrogen fixation by pea, faba bean, common bean and soybean plants, Soil Biol. Biochem. 26, 371–376.
- Deneyschen T., Strijdom B.W., Law I.J. (1998) Comparison of South African legume inoculants manufactured in 1978-79 and in 1995-97, in: Dakora F.D. (Ed.), Biological Nitrogen Fixation in Africa: Linking Process to Progress, African Association for Biological Nitrogen Fixation, Cape Town, South Africa, pp. 50–51.
- Devries J.D., Bennett J.M., Albrecht S.L., Boote K.J. (1989) Water relations, nitrogenase activity and root development of three grain legumes in response to soil water deficits, Field Crops Res. 21, 215–226.
- Dowlig D.N., Broughton W.J. (1986) Competition for nodulation of legumes, Annu. Rev. Microbiol. 40, 131–157.
- Duong T.P., Diep C.N., Khiem N.T., Hiep N H., Toi N.V., Lich N.V., Nhan L.T.K. (1984) *Rhizobium* inoculant for soybean [*Glycine max* (L.) Merrill] in Mekong Delta. II. Response of soybean to chemical nitrogen fertilizer and *Rhizobium* inoculation, Plant Soil 79, 241–247.
- Eaglesham A.R.J. (1989) Nitrate inhibition of root nodule symbiosis in doubly rooted soybean plants, Crop Sci. 29, 115–119.
- El-Hamdaoui, E., M. Redondo-Nieto, R. Rivilla, I. Bonilla, L. Bolanõs. (2003) Effects of boron and calcium nutrition on the establishment of the *Rhizobium leguminosarum* -pea

(*Pisum sativum*) symbiosis and nodule development under salt stress, Plant Cell Environ. 26, 1003–1011.

- Gates C.T., Muller W.F. (1979) Nodule and plant development in the soybean *Glycine max* (L.) Merr.: growth response to Nitrogen, Phosphorus and Sulfur, Aust. J. Bot. 27, 203–215.
- Giller K.E. (2001) Nitrogen Fixation in Tropical Cropping Systems, 2nd ed., CABI Publishing, Wallingford, Oxford.
- Giller K.E., McDonagh J.F., Cadisch G. (1994) Can biological nitrogen fixation sustain agriculture in the tropics? in: Syers J.K., Rimmer D.L. (Eds.), Soil science and sustainable land management in the tropics, CABI, Wallingford, UK, pp. 173–191.
- Graham P.H., Draeger K., Ferrey M.L., Conroy M.J., Hammer B.E., Martinez E., Naarons S.R., Quinto C. (1994) Acid pH tolerance in strains of *Rhizobium* and *Bradyrhizobium*, and initial studies on the basis for acid tolerance of *Rhizobium* tropici UMR1899, Can. J. Microbiol. 40, 198–207.
- Hardarson G., Bunning S, Montanez A., Roy R., MacMillan A. (2003) The value of symbiotic nitrogen fixation by grain legumes in comparison to the cost of nitrogen fertilizer used in developing countries, in: Hardarson G., Broughton W. (Eds.), Maximizing the use of biological nitrogen fixation in agriculture, Kluwer, pp. 213–220.
- Hungria M., Franchini J.C., Campo R.J., Crispino C.C., Moraes J.Z., Sibaldelli R.N.R., Mends I.C., Arihara J. (2006) Nitrogen nutrition of soybean in Brazil: contributions of biological N<sub>2</sub>-fixation and N fertilizer to grain yield, Can. J. Plant Sci. 86, 927–939.
- Hunt P.J., Wollum A.G., Matheny T.A. (1981) Effects of soil water on *Rhizobium japonicum* infection, nitrogen accumulation and yield in bragg soybean, Agric. J. 73, 501–505.
- IFDC (International Fertilizer Development Center) (2008) Soaring fertiliser prices threaten world's poorest farmers. Online at: http://ifdc.org/New\_Layout/News\_PressReleases/ index.html.
- Imsande J. (1986) Inhibition of nodule development in soybean by nitrate or reduced nitrogen, J. Exp. Bot. 37, 348–355.
- Isol T., Yoshida S. (1990) Growth, nodulation and nitrogen fixation of soybean plants with seeds treated with kasungamycin, Soil Sci. Plant Nutr. 36, 283–288.
- Java B., Everett R., O'Dell T., Lambert S. (1995) Legume seeding trials in a forested area of north-central Washington, Tree Planters' Notes 46, 19–27.
- Jayne T., Boughton D. (2006) Strategic option for achieving CAADP's agricultural growth targets, Paper presented at the SAKSS-SA regional workshop, Johannesburg, October 4th 2006.
- Johnson P., Zitsanza N., Sukume C., Rukuni M. (1994) The Oilseed Subsector in Zimbabwe, A Study for the World Bank (mimeo).
- Kaizzi C.K. (2002) The potential benefit of green manures and inorganic fertilizers in cereal production on contrasting soils in eastern Uganda, Ecology and Development Ser. 4, Cuviller Verlag, Gottingen, Germany.
- Kannaiyan S. (1993) Nitrogen contribution by Azolla to rice crop, Proc. Indian Natl. Sci. Acad. Part B Biol. Sci. 59, 309–314.
- Khonje D.J. (1989) Adoption of *Rhizobium* inoculation technology for pasture improvement in sub-Saharan Africa,

Department of Agricultural Research, Chitedze Agricultural Research Station, Lilongwe, Malawi, p. 14.

- Kipkoech A.K., Okiror M.A., Okalebo J.R., Martin H.K. (2007) Production efficiency and economic potential of different soil fertility management strategies among groundnut farmers of Kenya, Sci. World J. 2, 15–21.
- Kirda C., Danso S.K.A., Zapata F. (1989) Temporal water stress effects on nodulation, nitrogen accumulation and growth of soybean, Plant Soil 120, 49–55.
- Kueneman E.A., Root W.R., Dashiel K.E., Hohenberg J. (1984) Breeding soybeans for the tropics capable of nodulating effectively with indigenous *Rhizobium* spp., Plant Soil 8, 387–396.
- Mafongoya P.L., Bationo A., Kihara J., Waswa B.S. (2007) Appropriate technologies to replenish soil fertility in Africa, in: Bationo A., Waswa B.S., Kihara J., Kimetu J. (Eds.), Advances in integrated soil fertility management in sub Saharan Africa: challenges and opportunities, Springer, Netherlands, pp. 29–44.
- Maingi J., Shisanya C., Gitonga N., Hornetz B. (2001) Nitrogen fixation by common bean (*Phaseolus vulgaris* L.) in pure and mixed stands in semi-arid south-east Kenya, Eur. J. Agron. 14, 11–12.
- Mallik M.A.B., Tesfai K. (1993) Pesticidal effect on soybeanrhizobia symbiosis, Plant Soil 85, 33–41.
- Manyong V.M., Otite O., Dashiell K. (1998) Spread of new soybean varieties in a traditional soybean growing area of Nigeria, in: Bezuneh T., Ouedraogo S., Menyonga J.M., Zongo J.-D., Ouedraogo M.(Eds.), Towards Sustainable Farming Systems in Sub-Saharan Africa, AAFSRET, Ouagadougo, pp. 151–162.
- Marufu L., Karanja N., Ryder M. (1995) Legume inoculant production and use in East and Southern Africa, Soil Biol. Biochem. 27, 735–738.
- Michiels J., Verreth C., Vanderleyden J. (1994) Effects of temperature stress on bean nodulating *Rhizobium* strains, Appl. Environ. Microbiol. 60, 1206–1212.
- Miller R.H., May S. (1991) Legume inoculation: Successes and failures, in: Keister D.L., Cregan P.B. (Eds.), The Rhizosphere and plant growth, Kluwer Academic Publishers, Boston, pp. 123–134.
- Morris M., Kelly V., Kopicki R., Byerlee D. (2007) Fertilizer use in African agriculture: Lessons learned and good practice guidelines, The World Bank, Washington DC.
- Mpepereki S., Javaheri F., Davis P., Giller K.E. (2000) Soybeans and sustainable agriculture: promiscuous soybeans in Southern Africa, Field Crops Res. 65, 137–149.
- Mugabe J. (1994) Research on biofertilizers: Kenya, Zimbabwe and Tanzania, Biotechnol. Dev. Monitor, 18, 9–10.
- Munevar F., Wollum A.G. (1982) Response of soybean plants to high root temperature as affected by plant cultivar and *Rhizobium* strain, Agron. J. 74, 138–142.
- Mushi C.S. (1997) Annual report bean breeding. National Bean Programme, Selian Agricultural Research Institute, Arusha, Tanzania.
- Musiyiwa K., Mpepereki S., Giller K.E. (2005a) Physiological diversity of rhizobia nodulating promiscuous soybean in Zimbabwean soils, Symbiosis 40, 97–107.
- Musiyiwa K., Mpepereki S., Giller K.E. (2005b) Symbiotic effectiveness and host ranges of indigenous soybean

varieties in Zimbabwean soils, Soil Biol. Biochem. 37, 1169–1176.

- Mvula A.B., Tembo H., Mwale M., Woomer P.L. (1996) Growth response by common bean (*Phaseolus* varieties) to inoculation with *Rhizobium leguminasarium* bv. Phaseoli, AABNF 1996 proceedings.
- Nangju D. (1980) Soybean response to indigenous rhizobia as influenced by cultivar origin, Agronomy 72, 403–406.
- Ndakidemi P.A., Dakora F.D., Nkonya E.M., Ringo D and Mansoor H. (2006) Yield and economic benefits of common bean (*Phaseolus vulgaris*) and soybean (*Glycine max*) inoculation in northern Tanzania, Austr. J. Exp. Agric. 46, 571–577.
- Nehring R., Vialou A., Erickson K., Sandretto C. (2008) Assessing Economic and Environmental Impacts of Ethanol Production on Fertilizer Use in Corn, Annual Meeting, February 2–6, Dallas, Texas 6736, Southern Agricultural Economics Association.
- Nkonya E., Pender J., Kaizzi C., Kato E., Mugarura S., Ssali H., Muwonge J. (2008) Linkages between land Management, land degradation and poverty in Africa: The case of Uganda, Washington D.C., IFPRI Research Report (forthcoming).
- Odame H. (1997). Biofertilizer in Kenya: Research, production and extension dilemmas, Biotechnol. Dev. Monitor 30, 20–23.
- Okogun J.A., Sanginga N. (2003) Can introduced and indigenous rhizobial strains compete for nodule formation by promiscuous soybean in the moist savanna agroecological zone of Nigeria? Biol. Fertil. Soils 38, 26–31.
- Osunde A.O., Gwam S., Bala A., Sanginga N., Okogun J.A. (2003) Responses to rhizobial inoculation by two promiscuous soybean cultivars in soils of the southern Guinea savanna of Nigeria, Biol. Fertil. Soils 37, 274–279.
- Parker C.A. (1977) Perspectives in biological dinitrogen fixation, in: Hardy R.W.F., Silver W.S. (Eds.), A Treatise on Dinitrogen fixation, Sect. 3, Willey (Inter-science), New York, USA.
- Paynel F., Lesuffleur F., Bigot J., Diquelou S., Cliquet J. B. (2008) A study of <sup>15</sup>N transfer between legumes and grasses, Agron. Sustain. Dev. 28, 281–290.
- Poshiwa X., Nyoka R., Chikumba N., Gatsi T., Chakoma I. (2006) A baseline socio-economic evaluation of an European Union funded project on use of Micro-Symbionts in agroforestry systems in Zimbabwe, J. Sustain. Dev. Africa, 186–196.
- Pulver E.L., Brockman F., Wien H.C. (1982) Nodulation of soybean cultivars with *Rhizobium* spp. and their response to inoculation with *R. japonicum*, Crop Sci. 22, 1065–1070.
- Ranga-Rao V., Ayanaba A., Eaglesham A.R.J., Thottappilly G. (1984) Effects of *Rhizobium* inoculation on field-grown soybeans in Western Nigeria and assessment of inoculum persistence during a two-year fallow, Trop. Agric. 62, 125–130.
- Ranga-Rao V., Thottapilly G., Ayanaba A. (1981) Studies on the persistence of introduced strains of *Rhizobium japonicum* in soil during fallow and effects on soybean growth and yield, in: BNF Technology for Tropical Agriculture, pp. 309–315.
- Rinnofner T., Friedel J.K., Kruijff de R., Pietsch G., Freyer B. (2008) Effect of catch crops on N dynamics and following crops in organic farming, Agron. Sustain. Dev. 28, 551–558.

- Ryan J.G., Spencer D.C. (2001) Future Challenges and Opportunities for Agricultural R&D in the Semi-arid Tropics, International Crops Research Institute for the Semi-Arid Tropics (ICRISAT), Andhra Pradesh, India, 79 p.
- Saint Macary H., Bernard P., Montange D., Tranchant J.P., Verniau S. (1986) Setting and diffusion of a production system for legume *Rhizobium* inoculants, Symbiosis 2, pp. 363–366.
- Sakala K.M. (1990) Response of beans (*Phaseolus vulgaris* L.) to inoculation and fertilizer application, in: Soil Productivity Research Program Annual Report, Zambia.
- Sanginga N., Okogun J.A., Vanlauwe B., Diels J., Carsky R.J., Dashiell K. (2001) Nitrogen contribution of promiscuous soybean in maize-based cropping systems, in: Sustaining Soil Fertility in West Africa, SSSA Special Publication 58, Madison, WI, pp. 157–177.
- Sanginga N., Thottappilly G., Dashiel K. (2000) Effectiveness of rhizobia nodulating recent promiscuous soybean selections in the moist savanna of Nigeria, Biol. Biochem. 32, 127–133.
- Serraj R., Sinclair T., Purcell L. (1999) Symbiotic N<sub>2</sub> fixation response to drought, J. Exp. Bot. 50, 143–155.
- Silver M.C. (1976) Symbiotic nitrogen fixation by groundnuts (*Arachis hypogea*) in a ferrallitic soil in Uganda, MSc. (Agric.) Thesis, Makerere University, Uganda.
- Silver M.C.R., Nkwiine C. (2007) A review of work on rhizosphere microbiota in Uganda, Afr. J. Ecol. 45, 20–26.
- Sinclair T.R., Muchow R.C., Bennet J.M., Hammond L.C. (1987) Relative sensitivity of nitrogen and biomass accumulation to drought in field-grown soybean, Agron. J. 79, 986–991.
- Singleton P.W., Tavares J.S. (1986) Inoculation response of legumes in relation to the number and ineffectiveness of indigenous *Rhizobium* populations, Appl. Environ. Microbiol. 51, 1013–1018.
- Sivestre P. (1970) Travaux de l' IRAT sur le soya, Ford Foundation Grain Legume Seminar, 22–26 June, Ibadan, Nigeria.
- Smaling, E.M.A., Roscoe, R., Lesschen, J.P., Bouwman, A.F., Comunello F. (2008) From forest to waste: Assessment of the Brazilian soybean chain, using nitrogen as a marker, Agric. Ecosyst. Environ. (in press).
- Strijdom B.W. (1998) South African studies on biological nitrogen-fixing systems and the exploitation of the nodule bacterium-legume symbiosis, South African J. Sci. 94, 11–23.
- Strijdom B.W., van Rensburg J. (1981) Effect of steam sterilization and gamma irradiation of peat on the quality of *Rhizobium* inoculants, Appl. Environ. Microbiol. 41, 1344–1347.
- Thies J.E., Singleton P.W., Bohlool B.B. (1991) Modeling symbiotic performance of introduced rhizobia in the field by use of indices of indigenous population size and nitrogen status of the soil, Appl. Environ. Microbiol. 57, 29–37.
- Tran Y.T. (2004) Response to and benefits of rhizobial inoculation of soybean in the south of Vietnam, http://www. Cropscience.org.au/icsc2004/poster/2/5/6/1103\_tranyt.htm.
- Tu J.C. (1981) Effect of salinity on *Rhizobium*-root hair interaction, nodulation and growth of soybean, Can. J. Plant Sci. 61, 231–239.

- Van Rensburg H., Strijdom B.W. (1969) Strains of *Rhizobium japnicum* and inoculant production in South Africa, Phyto-phylactica 1, 201–204.
- Weaver R.W., Frederick L.R. (1974) Effect of inoculum rate on competitive nodulation of *Glycine max* L. Merrill. II. Field studies, Agron. J. 66, 233–236.
- Woomer P.L., Karanja N.K., Mekki E.I., Mwakalombe B., Tembo H., Nyika M., Silver M., Nkwiine C., Ndakidemi P., Msumali G. (1997), Indigenous rhizobia, legume response to inoculation and farmer awareness of inoculants in East and Southern Africa, in: Africa Crop Science Conference Proceedings, Vol. 3.
- Woomer P.L., Karanja N.K., Okalebo J.R. (1999) Opportunities for improving integrated nutrient management by smallhold farmers in the Central Highlands of Kenya, Afr. Crop Sci. J. 4, 441–454.
- World Bank (2008) World Development Report: Agriculture for Development, World Bank, Washington D.C.
- Yanggen D., Kelly V., Reardon T., Naseem A. (1998) Incentives for fertilizer use in sub-Saharan Africa: A review of empirical evidence on fertilizer response and profitability, MSU International Development Working Paper No. 70, Department

of Agricultural Economics, Michigan State University, East Lansing, USA.

- Zahran H.H. (1999) *Rhizobium*-legume symbiosis and nitrogen fixation under severe conditions and in an arid climate, Microbiol. Molec. Biol. Rev. 63, 968–989.
- Zaman-Allah M., Sifi B., L'taief M., El aouni H., Drevon J.J. (2007) Rhizobial inoculation and P fertilization response in common bean (*Phaseolus vulgaris*) under glasshouse and field conditions, Exp. Agric. 43, 67–77.
- Zengeni R., Giller K.E. (2007) Effectiveness of indigenous soybean rhizobial isolates to fix nitrogen under field conditions of Zimbabwe, Symbiosis 43, 129–135.
- Zhu Y., Sheaffer C.C., Vance C.P., Graham P.H., Russelle M.P., Montealegre C.M. (1998) Inoculation and nitrogen affect herbage and symbiotic properties of annual Medicago species, Agron. J. 90, 781–786.
- Zotarelli L., Resende A., Torres E., Hungria M., Urquiaga S., Boddey R.M., Alves B.J.R. (1998) O papel da soja para as reservas de N de um Latossolo Roxo sob plantio directo na regiao de Londrina, PR, in: Reuniao Brasileira de Fertilidade do Solo e Nutricao de Plantas, 23, Lavras.

# Conventionalisation of Organic Farming Practices: From Structural Criteria Towards an Assessment Based on Organic Principles

Ika Darnhofer, Thomas Lindenthal, Ruth Bartel-Kratochvil, and Werner Zollitsch

#### Contents

1	Intro	luction	332
2	An O	verview of the Conventionalisation Debate	333
	2.1	Core Issues of the Debate	333
	2.2	Conventionalisation in the European Context	335
	2.3	Shortcomings of the Debate Surrounding the	
		Conventionalisation Hypothesis	335
3	A Mo	re Discerning Approach to Studying Change	
	in Or	ganic Farming	337
4	Towa	rds Principle-Based Indicators	
	of Co	nventionalisation	339
	4.1	Organic Farming as Value-Based Agriculture	339
	4.2	Exploring Options for Building an Assessment	
		Framework	339
	4.3	Examples of Potential Indicators	
		for Conventionalisation	341
5	Concl	usion	345
Re	ference	es	346

Abstract Organic farming is widely perceived as being more environmentally friendly than conventional farming. As a form of sustainable agriculture, it receives substantial support from policy for its contribution to environmental protection as well as the provision of amenities such as biodiversity and cultural landscapes. Consumers are attracted to organic foods as they are produced without synthetic chemicals and comply with higher animal welfare standards. Although organic farming certainly has the potential to fulfil these expectations, studies have shown that some certified organic farms do not. Their practices comply with the regulations, but not with the principles of organic farming. This trend has been called 'conventionalisation' of organic farming. In this paper we review the studies that discuss the conventionalisation of organic farming, focusing on the farm level and on evidence from Europe. We argue that to strengthen organic farming's transformative potential, the debate must move beyond its focus on the bifurcation between artisanal and conventionalised organic farms, so as to capture the full range of empirical heterogeneity. Our core argument is that to adequately understand the dynamics within organic farming and their potential impact on the ability of organic farming to fulfil the expectations of consumers and policy-makers, it is not sufficient to focus on structural changes. Instead, we

I. Darnhofer (⊠)

E. Lichtfouse et al. (eds.), Sustainable Agriculture Volume 2, DOI 10.1007/978-94-007-0394-0\_18,
Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences

from Darnhofer et al., Agron. Sustain. Dev. 30 (2010) 67-81. DOI: 10.1051/agro:2009011

BOKU - Univ. of Natural Resources and Applied Life Sciences, Vienna, Feistmantelstr. 4, 1180 Vienna, Austria e-mail: ika.darnhofer@boku.ac.at

need to assess whether or not the observed changes comply with the principles and values that are the fundament of organic farming.

**Keywords** Organic agriculture • Conventionalisation debate • Europe • Plant production • Animal husbandry

#### 1 Introduction

Organic farming was developed in the 1940s in Switzerland by Hans Müller, Maria Biegler and Hans Peter Rusch, as well as in the United Kingdom by Lady Eve Balfour and Sir Albert Howard. Their developments were based in part on the concepts of biodynamic farming initiated in the 1920s by Rudolf Steiner in Germany. However, it was only in the 1980s, when the negative impact of intensive conventional agricultural methods became apparent, that interest in organic farming reached a broader public. In the European Union (EU) organic farming has received policy support since the early 1990s for its potential to contribute to environmental protection, rural development and animal welfare (EC, 2004; Häring et al., 2004; Nieberg et al., 2007). Organic farming is also linked to the production of quality food, and policies have been implemented to ensure transparency and fair competition on the market. Support by consumers and by policy-makers has resulted in a sizeable uptake of organic farming methods. In 2005, around 4% of the Utilised Agricultural Area of the 25 Member States of the EU was certified organic, representing over 6 million ha and nearly 158000 organic producers (Eurostat, 2007). In the last decade the market for organic products has grown steadily, both in Europe and elsewhere, and is expected to keep increasing (Michelsen et al., 1999; Willer et al., 2008).

Research on agronomic aspects of organic farming has shown that crop yields tend to be lower in organic farming (Mäder et al., 2002; Kaut et al., 2008), albeit less variable and less susceptible to drought (Lotter, 2003). Organic farming methods tend to increase soil organic matter and thus enhance soil fertility (Langmeier et al., 2002; Mäder et al., 2002), making it less dependent on external inputs. Organic farming may also provide ecosystem services, increase biodiversity and have a positive impact on the landscape (Letourneau and Bothwell, 2008; Norton et al., 2009). However, studies have pointed out that the potential benefits of organic farming are not always realised in all places and under all management systems (e.g., Trewavas, 2001; Rigby and Cáceres, 2001; Degré et al., 2007; Letourneau and Bothwell, 2008).

In the social sciences, there have been numerous studies on reasons for farmers to convert to organic farming (e.g., Fairweather, 1999; Padel, 2008) and for consumers to purchase organic foods (e.g., Brand, 2006; Holt, 2006). Research has also addressed the potential of organic farming to contribute to endogenous rural development, through fostering direct producer-consumer relationships (e.g., Moore, 2008; Renting et al., 2008; DuPuis and Gillon, 2009) or through service provision (e.g., Darnhofer, 2005; Schermer, 2006; Frederiksen and Langer, 2008; Lobley et al., 2009). Finally, a number of researchers have analysed the influence of legal regulations as well as of the entry of agribusiness and supermarkets into the organic food chain (e.g., Allen and Kovach, 2000; Alrøe and Noe, 2008; Tomlinson, 2008). However, as Lamine and Bellon (2009) have shown, there has been little dialogue between the agricultural and social sciences as well as a general lack of emphasis on the trajectories of organic farms and the understanding of transitions.

Transitions in organic agriculture and the trajectories of organic farms play a central role in the debate surrounding the conventionalisation hypothesis. The hypothesis was first put forward by Buck, Getz and Guthman in 1997 and has led to an on-going, intense debate on how the developments within organic farming can be understood and which patterns can be discerned. The debate essentially hinges on two aspects: whether the developments observed in the organic vegetable sector in California are universal and whether they are inevitable.

According to the conventionalisation hypothesis, organic farming is becoming a slightly modified version of modern conventional agriculture, replicating the same history, resulting in many of the same social, technical and economic characteristics (Buck et al., 1997; Hall and Mogyorody, 2001; Guthman, 2004a). Organic farming could thus be subjected to 'industrialisation', i.e. the implementation of economies of scale at the farm level (larger farms), increased reliance on purchased non-farm inputs (machinery, fertilisers, feed, agrichemicals), resource substitution (capital for land and labour), implementation of organisational

features associated with the concept of the 'firm', and mechanisation of the production process (Bowler, 1992). These mechanisms have been observed both at the farm level as well as in processing and marketing of organic food (Guthman, 2004a).

Conventionalisation is widely seen as problematic since organic farming has received public support for its potential to contribute to environmental protection and rural development (CEC, 2004). However, if organic farming increasingly comes to resemble conventional farming, this potential contribution is jeopardised, and organic farming might lose the support it currently receives from both consumers and policy-makers.

In this paper we present a brief review of the debate that has surrounded the conventionalisation hypothesis. We also discuss the empirical evidence that has been gathered to support and modify this hypothesis. We focus on the contributions from the EU<sup>1</sup> as the Common Agricultural Policy and the multifunctional model of agriculture are likely to offer different opportunities and constraints to organic farms than liberal agricultural policies. We will argue that a number of the 'symptoms', that have been used to identify conventionalisation at the farm level, may not be reliable indicators and that they are thus not well suited to achieve a comprehensive assessment of the type and direction of changes in organic farming. To overcome these weaknesses, we suggest that it may be necessary to design an assessment framework that is explicitly based on the ethical principles and values that are the foundation of organic farming. Such an assessment framework could be used both to assess the 'level of conventionalisation' on individual organic farms and to guide their development in accordance with the principles.

In this paper we will focus on conventionalisation at the farm level rather than taking an agrarian political economy approach. This in no way denies the importance of sophisticated, comprehensive studies of food networks (e.g. Goodman, 1999; Green and Foster, 2005; Brand, 2006; Lockie et al., 2006; Follett, 2009), nor does it indicate that we are not aware of the fact that all farmers operate within a larger political and economic framework that affects their agronomical practices (e.g., Allen and Kovach, 2000; DeLind, 2000; Guthman, 2004b; Thomas and Groß, 2005; Obach, 2007; Alrøe and Noe, 2008; Tomlinson, 2008). We also fully acknowledge the importance of consumer motivations (Lockie et al., 2002; Bähr et al., 2004; Padel and Foster, 2005; Holt, 2006) and the role supermarkets can play (Burch and Lawrence, 2005; Konefal et al., 2005) in the development of organic farming. However, we argue that if the imperative of agricultural intensification undermines the potential of organic farming to contribute to agricultural sustainability, then there should be clear evidence at the farm level. Also, as organic quality is mostly based on the production process, it is in large part defined by what happens at the farm level. Arguably, conventionalisation should lead to tangible changes in farm practices, which would threaten the core identity of organic farming. The farm is thus probably the most important level mediating the various influences exerted by the different scales in the hierarchies of agri-food networks.

## 2 An Overview of the Conventionalisation Debate

#### 2.1 Core Issues of the Debate

Trends towards conventionalisation were first reported from California (Buck et al., 1997; Guthman, 2004a), where high-value crops within the organic vegetable commodity chain were being appropriated by conventionally-based agribusiness. Many of these commercial farms were abandoning the more sustainable agronomic and marketing practices associated with organic agriculture (Guthman, 2000, 2004a). A first thread of the conventionalisation debate thus tries to assess whether such developments can also be identified in other locations and if so, how widespread such developments are. The relevant evidence from international studies has been reviewed by Lockie et al. (2006). They conclude that case studies from around the world, e.g. by Hall and Mogyorody, 2001; Lockie and Halpin, 2005; Jordan et al., 2006, show that concentration, de-localisation, institutionalisation and input substitution certainly are occurring to a significant extent (Lockie et al., 2006).

<sup>&</sup>lt;sup>1</sup> Especially in the 15 Member States of the former EU-15, as the Member States that joined the EU since 2004 tend to have a different agricultural structure and heritage, stemming from past socialist policies.



**Fig. 1** Schematic development of organic farming from the pioneers in bio-dynamic and bio-organic farming in the 1940s towards certified organic farming in the 1990s and expected further

development according to the conventionalisation and the bifurcation hypotheses. (Note: the graphic is indicative of trends and is not based on empirical data.)

A second thread of the conventionalisation debate hinges on whether conventionalisation affects all farms equally, or whether there might be a 'bifurcation' of the organic sector (see Fig. 1). A bifurcation would result if conventionalisation primarily affects large operations that specialise in mass-producing a few highgrowth, high-profit crops (Coombes and Campbell, 1998). The smaller, 'artisanal' farms continue to implement diversification strategies, using artisanal methods to grow a variety of marketable crops (Buck et al., 1997; Coombes and Campbell, 1998). This thread of the debate focuses on the ability of these 'artisanal' organic farms to resist the economic pressures exerted by the large operations, i.e. their ability to survive in the long term and ensuring that conventionalisation does not spread to all organic farms.

Guthman (2004b) has argued that agribusiness involvement unleashes the logic of intensification and therefore alters the conditions under which all organic growers operate. Through their control over processing and marketing, and through their introduction of industrial inputs, agribusinesses make the smaller operations less profitable, as they compete directly with the larger producers on the same markets. This puts pressure on the artisanal organic farmers to adopt conventional cropping, labour and marketing practices if they are to survive. The growing constraints in decision-making, coupled with increases in the economic pressures farmers face, could thus cause an erosion of the ethical attitudes and behaviours of farmers (Hendrickson and James, 2005).

Other authors, although acknowledging these pressures, have emphasised the smallholder's ability to resist concentration and specialisation and have pointed out that a number of factors countervail conventionalisation (Coombes and Campbell, 1998; Lockie et al., 2006; Guptill, 2009). These include the technology barriers faced by larger and more monocultural operations; the biological limits to input substitution; the ability of household-based enterprises to cope with unfavourable returns through self-exploitation and reduced consumption; the competitive advantage of small enterprises in a range of markets and under a range of policy conditions; their ability to target market niches; as well as consumer demands for what are perceived to be local, speciality products. Coombes and Campbell (1998), in their analysis of the development of organic farming in New Zealand, not only emphasise that smallholders are expected to survive, but also show that the relationship between small and large growers may well be complementary. Indeed, larger actors may initiate research and market development, and/or focus on different products or markets.

The various contributions to the conventionalisation debate have thus shown that although symptoms of conventionalisation can be identified fairly easily, a widespread conventionalisation of organic farming is (still) constrained. This conclusion needs to be understood as being based on case studies in countries with a liberalised agricultural policy, e.g. California (USA), Ontario (Canada), New Zealand and Australia. In such a context, farmers receive few or no subsidies and are thus more likely to be affected by the vagaries of market forces. However, as Guthman (2004b) points out, the type and degree of state support along with the agrarian structures on which organic farming was built are likely to have a strong influence on the extent, severity and pervasiveness of conventionalisation.

## 2.2 Conventionalisation in the European Context

In the EU, agriculture is characterised by a vast majority of family farms and a minority of corporate farms. The Common Agricultural Policy has embraced the concept of multifunctional agriculture, thus recognising that farms not only produce food, but also provide services such as landscape amenities, recreational space, environmental protection and preservation of cultural heritage. Whereas large-scale agriculture may be efficient in producing food and fibre, small-scale farms tend to be more efficient in supplying services that are valued at a local level. Farms thus receive direct payments for the provision of public goods. This supportive policy environment creates a different set of opportunities for organic farmers.

Still, case studies assessing the changes in organic farming in Europe report on trends and practices that could indicate conventionalisation processes. For example, the size of organic farms is increasing, e.g. in Denmark (Langer and Frederiksen, 2005) and in Germany (Best, 2008). An analysis of dairy herds in Norway shows that later entrants tend to have a higher level of registered disease treatments per cow, and an intensification of milk production based on a higher use of concentrates (Flaten et al., 2006). Smith and Marsden (2004) point out that the over-supply of some organic products (e.g., milk) has led to a 'farm-based cost-price squeeze' in the UK, which might be forcing farmers into progressively more intensive production strategies. Best (2008) finds that later entrants seem to be somewhat less concerned about the environment. He also reports fewer mixed farms and a drop in direct marketing. De Wit and Verhoog (2007) report conventionalisation trends in organic pig and poultry production in the Netherlands. Conventionalisation has also been reported from arable farming, where permitted fertilisers of conventional origin are increasingly used (e.g. vinasse, a byproduct of the sugar beet industry). The intensive use of fertiliser has resulted in mineral surpluses in the soil and higher nitrate levels, e.g. in organic carrots (De Wit and Verhoog, 2007; Padel et al., 2007). Despite these symptoms of conventionalisation, a Europe-wide study concludes that conventionalisation does not (yet) seem to be a dominant phenomenon in organic farming as a whole (De Wit and Verhoog, 2007).

Thus, in Europe as elsewhere, some organic farms are implementing practices that may not be sustainable but that are not explicitly prohibited by the standards (Padel et al., 2007). Certified organic farming by itself is thus no guarantee for its alternativeness, as has been pointed out by Guthman (2004b). The question thus arises whether the symptoms that have been identified indicate that conventionalisation is progressing, thus undermining the potential contribution of organic farming to quality food and to environmental protection. To answer this question it is not only necessary to assess whether the identified practices are spreading, it is also necessary to assess whether the practices and symptoms reported in the studies are valid and reliably indicate the conventionalisation of organic farming.

## 2.3 Shortcomings of the Debate Surrounding the Conventionalisation Hypothesis

When critically assessing the methods used in the various studies, a range of weaknesses can be identified. Most of these are due to the fact that the available data does not allow to sufficiently capture the heterogeneity and complexity of the on-going processes within organic farming. Indeed, there has been a focus on analysing aggregated statistical data, and there has been a lack of distinction between farm types, e.g. part-time vs. full-time farmers or corporate vs. family farms. Furthermore, there is a lack of statistical time-series data to ascertain long-term trends. In this section we review some of the approaches leading to conclusions that are based on circumstantial evidence rather than on valid variables and rigorous analysis of comprehensive data sets.

One of the methodological weaknesses lies in comparing early converters with later converters and deriving conclusions on the change in attitudes, values and practices of organic farmers. In this approach, a small group of pioneers who have been organic for an extended period of time are compared with farmers who have limited experience with organic farming as they converted recently. This comparison is fundamentally problematic, as it tends to ignore the learning processes leading to change in knowledge and attitudes that farmers undergo after conversion (Padel, 2008). To reliably assess changes in attitudes, a longitudinal study of both early and later converters would be required, but none has been reported so far.

A similar weakness can be found in studies that analyse aggregated statistical data and assess changes in the 'average' organic farm at two points in time. These studies often conclude that organic farms are now larger than they used to be, or that they are more specialised than they used to be. These changes are then assumed to derive from farmers' preferences, indicating conventionalisation of on-farm practices. The problem with this approach is that the influence of changes in the wider environment (markets, policies, technologies) tends to be under-theorised. The identified differences may thus be the result of processes that are not linked to conventionalisation. For example, a reduction in the share of organic farms involved in animal keeping might be due to the spread of organic farming in a new agro-ecological environment, i.e. a new cohort of organic farms, rather than different practices by the same cohort. For instance, in Austria most organic farms in the period 1995-2000 were grassland-based dairy farms. After the year 2000 arable farms started converting to organic farming, but these had few, if any, animals even prior to conversion. Similar shifts in the national composition of organic farms were also reported from Denmark (Langer, 2002). Thus, data indicating changes in the 'average' organic farm must be interpreted carefully, so as not to infer unwarranted causal links.

Unwarranted conclusions can also be the result of a lack of distinction between farm types, e.g. because the survey sample is not large enough or because the official statistics do not contain the necessary variables. However, capturing the heterogeneity of organic farms is important as it is likely that different farm types, e.g. full-time vs. part-time farmers, family farms vs. corporate farms, mixed farms vs. grassland-based dairy farms (which are often found in less favoured areas), have different options to face market pressures and price squeezes. Indeed, as part-time farmers derive most of their household income from off-farm work, they are not affected by market pressures to the same extent as corporate farms (Best, 2008). This type of distinction is important, as in many European countries, a large share of farms are managed parttime: in the sample surveyed by Best (2008) 68% were part-time farmers; in Austria 61% of all farmers are part-time farmers (BMLFUW, 2007). To be able to accurately capture conventionalisation would require an analysis differentiating between, e.g., farm types, commodities and marketing channels. Only then will it be possible to do justice to the complexity of organic production (Sylvander et al., 2006; Rosin and Campbell, 2009; Lamine and Bellon, 2009).

The lack of an adequate, nuanced analysis has also been raised by Lockie and Halpin (2005, p. 287), who point out that the binary opposition between 'artisanal' and 'industrial', between 'lifestyle' and 'agribusiness' producers, can be problematic, as it "conflates differences in economic scale with differences in production practices, market relationships and grower motivations that have not been empirically verified". They thus point out the need to "unpack the concept of conventionalisation and avoid the uncritical aggregation of multiple dualisms between small and large, artisanal and industrial, radical and regulatory, local and international, regenerative and substitutionist, and so on through the related concept of bifurcation" (Lockie and Halpin 2005, p. 304).

At a more general level, the conventionalisation debate also suffers from resting on implicit and poorly justified stereotypes. As Langer and Frederiksen (2005) have pointed out, conventionalisation builds on the notion that until a decade or two ago organic farms were smaller, less specialised and less intensive than conventional farms, and that this might now be changing. This assertion regarding 'early' organic farms is often made, although there is little (if any) historical data on organic farms. We thus do not have sufficient data to ascertain whether they all were mixed farms or whether there has always been a certain level of heterogeneity. The stereotype of the 'small' organic farm is also surprising, as in most European countries organic farms are larger than conventional farms: the average size of organic holdings in the EU-25 in 2005 was 38.7 ha, compared with 16.0 ha for all holdings (Eurostat, 2007). In this context, an increase in farm size is hardly an indication of a convergence between organic and conventional farms. Thus, whereas conventionalisation has been linked to increased farm sizes, the converse need not be true: not every instance of scale increase necessarily indicates conventionalisation.

The conventionalisation debate may also have suffered from being mostly conducted by social scientists. Due to the dearth of interdisciplinary studies (Watson et al., 2008; Lamine and Bellon, 2009), the links between the changes noted by social scientists and the agronomic practices of farmers have not been established. In other words, the implied effect of the changes within organic farming on variables indicating environmental protection, sustainable plant production, animal welfare and soil fertility is rarely based on empirical evidence. For example, a reduction of mixed farms and of animal keeping is hypothesised to indicate a stronger reliance on off-farm fertiliser and thus input substitution. However, before reaching this conclusion, it would be important to assess whether the number of animals kept in the first place was sufficient to cover the nutrient needs of the field crops. Indeed, from an agronomic point of view it would be hard to argue that 100 growing-fattening pigs make a sizeable contribution to nutrient cycling on a 50-ha crop farm. Thus, if the conventionalisation debate is to be relevant to practitioners and inform them about the development options within organic farming, then the debate will need to integrate other disciplines into the discourse, e.g., crop and livestock production scientists as well as soil scientists.

Overall, we conclude that despite the case studies reporting symptoms that have been linked to it, the available data is inadequate to confirm or to refute the conventionalisation hypothesis in the European context. This is not least due to the fact that the variables used to identify the changes do not reliably indicate conventionalisation and that the available data does not allow for the nuanced analysis required to adequately capture the heterogeneity of organic farms or the complexity of the change dynamics.

## 3 A More Discerning Approach to Studying Change in Organic Farming

The review of the debate surrounding the conventionalisation hypothesis has shown that there are changes within organic farming. These transition dynamics are sometimes - if implicitly - seen as problematic in and of themselves. In this view, the pioneers are understood as the original, truest proponents of organic farming. The goal of organic farming is thus to replicate their practices, now and in the future. The original practices should not be changed or modified as it would undermine the nature of organic farming. This view presents conventionalisation as problematic, primarily because it implies a departure from the practices of the pioneers. This justification for opposing conventionalisation is questionable, if there is a lack of awareness that the converse is not necessarily the case. Indeed, although conventionalisation is a departure from the practices of the pioneers, not every departure from the practices of the pioneers is an indication of conventionalisation. We thus should no longer focus on whether organic farming is changing or not: there is ample evidence that it is. It would be more fruitful to focus on how organic farming is changing, taking care to capture the whole range of changes.

Indeed, organic farming is not a Luddite movement that seeks to turn back the clock. Organic farming should not be limited to the practices and methods of the pioneers. Not least because some of their practices, especially regarding animal welfare, were questionable (Sundrum, 2005). Rather, organic farming needs to be understood as dynamic, i.e. a system that responds to internal and external demands and conditions (IFOAM, 2005). Organic farming is thus involved in an on-going process of reviewing existing methods, assessing new technological developments, e.g. plant breeding, soil management and tillage, mechanical weed management, and implementing relevant insights from agro-ecological research (Niggli, 2007). Looking at the 1940s, it is clear that the pioneers also understood organic farming as dynamic and did not have qualms adjusting those aspects that they found unsuitable. For example, the bio-organic farming as developed by Müller and Rush is no longer based on anthroposophy, as was (and still is) the case of biodynamic farming (Aeberhard and Rist, 2009). Instead



Fig. 2 Schematic representation of changes in the relative importance of values, natural sciences and economics for production decisions on organic farms

of using antrophosophy as their guideline, Müller and Rush have given agro-ecology and scientific thinking a much larger role (Fig. 2).

Changes are thus not problematic in and of themselves and organic farming has benefited from them in the past. Moreover, change is necessary for the survival of organic farming. Indeed, farmers now face an environment that is very different from the one faced by the pioneers: they have more regulatory constraints, but access to better developed markets; they face higher competition, but have more scientific and practical advice available to them; they face rapid changes, but have better communication possibilities. Both the constraints and the opportunities open to organic farmers are different, and as the farmers tackle these demands, it is bound to reflect on their on-farm practices.

What needs more attention is the diversity and type of changes taking place. Darnhofer (2006) has proposed to distinguish between first-order changes, i.e. changes that do not undermine the principles of organic farming (what may be termed 'professionalisation') and second-order changes. Second-order changes entail a discontinuity, they involve a shift in the rules that govern on-farm decisions. Conventionalisation can be labeled a second-order change, as the organic principles are sidelined in favour of economic profitability. Although organic farmers have always had to ensure that their farm is economically sustainable, within conventionalisation, economic considerations tend to marginalise both the principles and agroecology (Fig. 2).

Moving towards a more discerning analysis of the changes taking place on certified organic farms, we propose to define 'conventionalisation' as the introduction of farming practices that undermine the principles of organic farming, i.e. a second-order change. The assessment of a move towards conventionalisation should thus be based on criteria and indicators showing that the principles of organic farming are being undermined, not on structural changes (that may or may not indicate an undermining of principles). This approach implies moving away from focusing on the two extremes of 'true organic' vs. 'conventionalised'. Instead of focusing on the extremes, it might be more fruitful for research to focus on the field in-between, on the range of possibilities within organic farming. The goal is to leave organic farming room to change while providing farmers with guidance on the development paths that are in line with the principles of organic farming. This seems more constructive than debating whether organic farming is being conventionalised or not.

#### 4 Towards Principle-Based Indicators of Conventionalisation

In this section we present a potential approach towards a principle-based indicator that would allow to distinguish between changes that follow the principles of organic farming and those that do not. We present an outline that will require substantial further development to be applicable empirically. Our goal is not to present a polished toolbox, but to indicate a way to overcome the weaknesses of previous approaches to assessing change in organic farming. These have proven too rough and thus could not capture the developments within organic farming with the required discernment.

## 4.1 Organic Farming as Value-Based Agriculture

Organic farming has set out to be an alternative to conventional agriculture and food chains. It is based on principles and values (Luttikholt, 2007; Besson, 2008). The International Federation of Organic Agriculture Movement (IFOAM), after a concerted and participatory process, has formulated four principles to inspire action (IFOAM, 2005):

- the principle of health: "Organic agriculture should sustain and enhance the health of soil, plant, animal and human as one and indivisible";
- the principle of ecology: "Organic agriculture should be based on living ecological systems and cycles, work with them, emulate them and help sustain them";
- the principle of fairness: "Organic agriculture should build on relationships that ensure fairness with regard to the common environment and life opportunities"; and
- the principle of care: "Organic agriculture should be managed in a precautionary and responsible manner to protect the health and well-being of current and future generations and the environment".

These principles have a strong ethical component and display a much wider view of agriculture compared with the Good Agricultural Practice which may serve as a guide to conventional farming (e.g. DARD, 2008). However, these values are only partially codified in rules and regulations, thereby allowing the compromising of a more holistic vision of organic farming (Milestad et al., 2008). Indeed, the organic standards tend to focus on values and practices that are easy to codify and audit through the inspection and certification process, such as what inputs are permitted or excluded (Lockie et al., 2006; Padel, 2007). It has been argued that the dominant regulatory focus on inputs is much more likely to encourage entrants who can substitute allowed materials for disallowed materials (Rosset and Altieri, 1997; Guthman, 1998; Michelsen, 2001). Thus, the necessity of transparency in the interest of trade has made possible a rationalisation and simplification of organic meanings (Tovey, 1997; Allen and Kovach, 2000). The new European Regulation for organic production (EC, 2007) does include principles for organic production. However, not all are translated into production rules that can be part of inspection and certification (Padel et al., 2007). This mostly affects agro-ecological system values such as bio-diversity and nutrient recycling, as well as the lack of social considerations (Padel, 2007; Lockie et al., 2006). However, as Padel (2007) points out, the fact that some core values are not part of the standards does not mean that they are less important to organic stakeholders.

The organic farming associations may thus be called upon to ensure that their members do not focus exclusively on implementing the minimum requirements necessary for certification. They may also need to counter the attractiveness of exploiting legal loopholes. The goal would be for the associations to uphold the impetus to keep developing on-farm practices that implement the principles in ever more comprehensive ways. To achieve this, an assessment framework whose indicators are based on the principles of organic farming would seem a useful tool. Although developing this framework in detail is beyond the scope of this paper, we would like to propose some preliminary concepts to concretise our proposal.

## 4.2 Exploring Options for Building an Assessment Framework

The goal of developing an assessment framework is for it to serve as a tool to assess and guide future



Fig. 3 Schematic structure of a tool to assess the extent to which an organic farm complies with the principles of organic farming

developments of organic farming methods and practices. We are aware of the practical and conceptual challenges involved in developing such an indicator set. Various authors have discussed the issues related to the necessity to cope with different units of measurement, as well as the trade-offs between complex index measurements and simplified approaches that can be understood by practitioners who do not have a background in statistics (e.g., Andreoli and Tellarini, 2000; Malkina-Pykh and Pykh, 2008; Meul et al., 2008). Despite these hurdles, a number of models for constructing an indicator-based framework to assess sustainability of farms have already been built (e.g., van Cauwenbergh et al., 2007; Knickel, 2008). As the goal is to derive indicators from the principles, we would suggest a hierarchical framework (see van Cauwenbergh et al., 2007). Figure 3 presents a concept of such a framework linking principles, criteria, indicators and reference values which are measured on different spatial scales (plot, farm, landscape). The first level of the hierarchy is made of the principles of organic farming, i.e., the general objectives to be achieved, which clearly go beyond the organic standards.

The second hierarchical level is the criterion, i.e. the resulting state or aspect of the farming system when its related principle is respected. Criteria are thus specific objectives or desirable trends. Criteria would need to encompass the environmental, economic and social functions of an organic farm. Criteria challenging the positivistic view of traditional science, such as subjective experiences, may also be considered (Lund and Röcklingsberg, 2001; Meul et al., 2008).

The third hierarchical level is the indicator. Indicators are variables of any type that are used to measure compliance with a criterion. A set of indicator values should provide a representative picture of the implementation level of the organic farming principles on a farm. The goal would thus be to ensure the selection of a core, coherent and consistent list of indicators. A number of the indicators should be site- and scalespecific as some criteria and indicators may not be relevant for all regions or all farm types. Some indicators could also be based on existing measurements, e.g. the animal needs index; the indexes used for selecting breeding animals for a number of different traits at the same time (Hazel, 1943); or indicators developed to assess the environmental performance at the farm level (Braband et al., 2003; Meul et al., 2008; Niemeijer and de Groot, 2008).

The fourth and lowest level of the framework is made up of the reference values which describe the desired level for each indicator. The reference values can be either an absolute value or a comparative value, e.g. based on similar organic farms in the same agroecological region. The absolute reference values might be either target values, i.e. desirable conditions, or they could be threshold values in the form of minimum or maximum levels, or a range of acceptable values.

We are aware of the contested and negotiated nature of indicators and reference values. It will be a challenge to legitimise indicators when faced with divergent claims made by different groups (Slee, 2007). The selection of the indicators and the reference values should thus be part of a participative process which includes a wide range of stakeholders (Knickel, 2008; van Ittersum et al., 2008). The discussions could be used as a tool to make the range of positions and reasoning transparent.

Such a principle-based assessment framework needs to ensure a holistic approach to organic farming, by at least being based on a whole-farm assessment. The framework needs to be based on an understanding of the farm as a complex organism, rather than the sum of more or less independent activities. The latter is a reductionistic approach that does not do justice to the systemic underpinning of organic farming. The goal would be to show the extent to which a farm implements the integrity approach, rather than the no-chemicals or the agro-ecology approach (Verhoog et al., 2003; Baars and Baars, 2007).

The framework itself would be expected to change over time, as the principles of organic farming are refined (Luttikholt, 2007), scientific knowledge increases or societal values and concerns evolve. The goal would be to ensure a coevolutionary process, between organic farming principles, practices of organic farmers, and technical and economic feasibility as well as societal expectations.

As the assessment is based at the field, at the farm and at the landscape level, it allows the links between management by the farmer and impacts and effects on the agro-ecosystem to be addressed explicitly. This would allow the assessments derived from the framework to be used by a variety of actors: organic farmers, organic farming associations, researchers and regional policy decisionmakers. Within the framework, farms could be compared within a regional best-practice assessment. Such a regional ranking would take into account what is feasible in the region and provide each farm with a nuanced and detailed profile of the areas in which it is performing well, and those criteria where improvements seem desirable and possible. The assessment would also be dynamic since it takes into account developments and improvements in best practices.

## 4.3 Examples of Potential Indicators for Conventionalisation

To illustrate the types of indicators that can be considered within the assessment framework, we have compiled some indicators for crop production (see Table 1) and animal husbandry (see Table 2). These indicators were selected focusing on those areas where practices threaten to undermine the principles. As stated in the previous section, we want to emphasise that any individual indicator is only meaningful when seen within the total constellation of a set of indicators (Niemeijer and de Groot, 2008). Indeed, an individual indicator cannot effectively capture a causal network, i.e., the whole range of causes and effects and their

Table 1 Examples of indicators that could be used to ident	ify conventionalisation in crop production, and related princi	ples of organic farming (IFOAM, 2005)
Indicators of conventionalisation	Justification and comments	Principle of organic farming <sup>1</sup>
Low share of legumes in the crop rotation.	Legumes are necessary for humus build-up and a key element of the nitrogen supply. As an indicative reference value, there should be at least 20% legumes, but this could be higher, depending on the agro-ecological specificities of a site.	Ecology – Production is to be based on ecological processes. Nourishment is achieved through the ecology of the specific production environment. In the case of crops this is based on soil fertility and diverse crop rotation (to build up resilience of the agro-ecological system).
High share of cereals in the crop rotation.	A high share of cereals tends to lead to plant diseases and pests, deficiencies in the nutrient supply and poor humus build-up. As an indicative reference value, cereals should not use more than 70% in the crop rotation.	Ecology – Organic agriculture should attain ecological balance and resilience (see above) through the design of cropping/farming systems.
Inadequate crop rotation or unbalanced crop sequence.	The incidence of (soil-borne) pests, diseases and weeds increases. Also, crop rotations are needed to balance the soil nutrient demands of various crops and to avoid soil depletion.	Ecology – Organic agriculture should be based on living agro-ecological systems and cycles, work with them, emulate them and help sustain them.
Reliance on easily soluble (nitrogen) fertilisers (e.g. vinasse).	High levels of easily soluble nutrients (esp. nitrogen) threaten plant health, lower product quality and lead to problematic NO <sub>3</sub> and N <sub>2</sub> O emissions.	Ecology – Production is to be based on ecological processes and recycling.
Prolonged and intensive use of plant protection products that are known to be problematic (e.g. sulphur, copper, pyrethrum).	These products are known to be toxic and to accumulate in the soil. They threaten beneficial insects and soil fertility.	Ecology and Health – Organic agriculture should avoid the use of fertilisers, pesticides, etc. that may have negative ecological and health effects.
Widespread use of practices that require a high level of external inputs (energy, fertilisers, feedstuffs, materials).	A farm (or a region) should aim at a balanced energy and nutrient budget and not be excessively dependent on external inputs. External inputs tend to be linked to high levels of CO <sub>2</sub> emissions and energy use.	Ecology – Inputs should be reduced by reuse, recycling and efficient management of materials and energy in order to maintain and improve environmental quality and conserve resources.
High share of varieties or cultivars which are not adapted to organic farming.	As these varieties are not adapted to the local agro-ecosystem and/or organic management they do not lead to stable yields appropriate to the local conditions. They usually require higher levels of external invuts.	Ecology – Organic management must be adapted to local conditions, ecology, culture and scale, in order to build up agro-ecological resilience.
Low level of biodiversity on the cropland.	The agro-ecological stability and the biodiversity are reduced. Cropland biodiversity includes the types of crops planted each year, and the number of cultivars used for a crop as well as the varieties used as green manure.	Ecology – Organic cropping is based on resilient agro-ecological systems, where biodiversity plays an important role. Organic agriculture should contribute to the maintenance of genetic and agricultural diversity.
Low level of biodiversity around the cropland (e.g., hedges, field margins).	Reduced agro-ecological stability (e.g. of beneficial organisms), poor protection of natural resources, lack of habitats and lack of contribution towards the cultural landscape.	Ecology – Organic farming should protect and benefit biodiversity (e.g. rare species), air and water. Role of biodiversity in resilience.
Few measures to actively protect and care for	Reduced agro-ecological stability; lack of habitats.	Ecology - Those who produce organic products should
conversion summer a cas on the ratin (c.g.; ponus, marshes).	ince areas are onen of ingle corrected value, out of low economic/productive value.	protect and ochecut the common current including landscapes [and] habitats.

These are examples, as most conventionalisation indicators are related to several principles and several aspects of the same principle.

<b>Table 2</b> Examples of indicators that could be used to i (IFOAM, 2005)	identify conventionalisation in animal husbandry and grasslar	nd management, and related principles of organic farming
Indicators of conventionalisation	Justification and comments	Principle of organic farming <sup>1</sup>
High incidence rate of metabolic disorders/Frequent	Metabolic disorders (e.g. acidosis or ketosis in	Health - Organic agriculture should sustain and enhance
veterinary treatments required.	ruminants, which are often referred to as 'production diseases') can be an indicator of high production intensity, inadequate animal housing conditions, or feeding.	the health of () animal[s]. Health is the wholeness and integrity of living systems. It is not simply the absence of illness.
High incidence rate of abnormal animal behaviour.	Atypical behaviour (e.g. feather pecking and cannibalism in laying hens) tends to indicate a high animal population density, inadequate housing system, a lack of appropriate stimuli, or lack of space to allow natural behaviour.	Fairness – Animals should be provided with the conditions and opportunities of life that accord with their physiology, natural behaviour and well-being.
Average life time of the animals is substantially below the average productive life span of the species.	In organic farming animals should not be conceived of as 'throw-away' animals. The focus should be on the potential productive life span of the species.	Fairness – Respect, justice and stewardship of the shared world () in their [the people's] relations to other living beings.
Use of breeds that are not appropriate for organic farming systems (e.g. poultry hybrids, breeds with extreme muscle conformation, dairy breeds with a predisposition for metabolic disorders).	Increases the incidence of diseases due to low robustness of the animals.	Ecology – Organic management must be adapted to local conditions, ecology, culture and scale.
Number of rare breeds kept on-farm.	The use of rare breeds is a valuable contribution to the maintenance of biodiversity. However, they are often linked to lower economic productivity and thus neglected.	Ecology – Organic agriculture should contribute to the maintenance of genetic and agricultural diversity.
High share of reproduction through artificial insemination. The sires used stem from the use of embryo transfer (ET bulls).	It would be preferable to keep male animals on the farm (or on a neighbouring farm) and allow the animals to mate naturally.	Fairness – Animals should be provided with the conditions and opportunities of life that accord with their physiology, natural behaviour and well-being.
High share of concentrates in the ration of herbivores (e.g. dairy cows).	This is more useful as an indicator than e.g. the milk production per cow, which can vary strongly depending on the genetic potential and on the agro-ecological conditions.	Fairness – Animals should be provided with the conditions and opportunities of life that accord with their physiology, natural behaviour and well-being.
High share of feed that is purchased (industrially produced) rather than produced on the farm (or by neighbouring farms).	The number of animals kept on-farm needs to be in relation to the feed that can be produced on the farm's own fields.	Ecology – Nourishment and well-being are achieved through the ecology of the specific production environment.
Restricted access to pastures (for e.g. cows).	Each species should be allowed to display typical behaviour (e.g. grazing for cows). Sufficient exercise also reduces the risk of lameness.	Fairness – Animals should be provided with the conditions and opportunities of life that accord with their physiology, natural behaviour and well-being.
		(continued)

Table 2 (continued)		
Indicators of conventionalisation	Justification and comments	Principle of organic farming <sup>1</sup>
Few structuring elements in the animal housing, including the outdoor run.	Structural elements are necessary for orientation and motivation for species-specific behaviour, e.g. separate resting and activity areas, perches for poultry.	Fairness – Animals should be provided with the conditions and opportunities of life that accord with their physiology, natural behaviour and well-being.
Type of storage, treatment and use of animal urine and manure (i.e. prevalence of slurry).	Whether urine and manure are collected and stored separately or as slurry has implications for their use (i.e. effect on the species composition of the grassland).	Ecology – Organic agriculture should be based on living ecological systems and cycles, work with them, emulate them and help sustain them.
Animal housing makes extensive use of non-renewable materials, such as plastic sheets, rubber mats, etc.	Use of ecologically sound design and of renewable materials would be preferable.	Ecology – Inputs should be reduced by reuse, recycling and efficient management of materials () to () conserve resources.
Low amounts of litter provided (i.e., animal housing that is geared towards litter minimised system).	The amount of litter should be adequate to fulfil the needs and natural behaviour of the animals.	Fairness – Animals should be provided with the conditions and opportunities of life that accord with their physiology, natural behaviour and well-being.
Low time budget for the care of the individual animal and for herd management.	Increasing herd sizes reduces the time budget per animal of the stockpersons. If this time budget decreases below a threshold (which needs to be defined separately for each livestock category), animal welfare suffers.	Care – Organic agriculture should be managed in a precautionary () manner.
Only a few animal types are kept on-farm.	It would be desirable for farmers to keep ruminants as well as monogastric livestock to integrate animal husbandry better with plant production and thereby balance demand and supply of feed resources within a farm.	Ecology – Inputs should be reduced by reuse, recycling and efficient management of materials and energy. Production is to be based on ecological processes, and recycling.
The choice of mowing dates of the grassland is not based on ecological criteria.	Grassland management needs to take into account ecological criteria to ensure the upkeep of the biodiversity in the grassland (flora and fauna), and the ecological stability of the grassland.	Ecology – Organic pastoral systems should fit the cycles and ecological balances in nature (role of biodiversity, s. a. and enhancement of rare species).
High incidence rate of broad-leaved dock on grassland.	Broad-leaved dock tends to be linked to a level of slurry application that is not adequate for the local conditions, as well as a high frequency and poor timing of mowing. This threatens the yield, fodder quality and the ecological stability of grassland.	Ecology – Organic management must be adapted to local conditions [and] ecology.
<sup>1</sup> These are examples, as most conventionalisation indicato	ors are related to several aspects of the same principle and sev	eral principles.

inter-relation. A set of indicators, however, should be able to identify the production logic underlying farm management.

Most of the indicators in Tables 1 and 2 are linked to finding "conventional" solutions to challenges faced in production and management decisions. They are mostly linked to a production logic that seeks to maximise production, i.e. aiming for yields above the level adapted to the local agro-ecological conditions, and that lacks an understanding of the interrelationships between different farm activities; and/or to short-term thinking that might be the result of economic pressures and the (perceived) need to 'cut corners', e.g., to reduce the share of legumes in the crop rotation, as legumes tend to have a low gross margin.

Indeed, a number of the indicators are linked to economic profitability, being a dominating decision criterion in conventionalised farming (Fig. 2). However, in organic farming short-term economic aspects need to be balanced with long-term ecological and social considerations to achieve sustainability. It reflects the approach at the core of organic farming of the ethical values of stewardship and moral care for the land and thus organic farmers "say no to using chemicals, not to (always) putting every acre under tillage, not to (always) striving for the largest yield" (Stock, 2007, p. 97).

A departure from the organic principles is also reflected in practices that do not seek systemic solutions to problems, but focus on the topical alleviation of a production problem. The farm is thus not understood as a whole where each part needs to be seen in the context of the other parts. Instead, each part of the farm is perceived as separate and problems appearing in that part are solved 'locally' (separately) without considering links to other activities on the farm.

For example, a high incidence of broad-leaved dock (Rumex obtusifolius) in permanent grassland can be an indicator of intensification. It tends to be found on farms where management is guided more by economic goals and technical objectives, sidelining ecological knowledge and locally adapted management. Problems with broad-leaved dock can be linked to the application of slurry, intensive grassland use (e.g. high frequency and poor timing of mowing), as well as poor grassland management, resulting in low plant cover and soil compaction (Dietl and Lehmann 2004; FiBL, 2006). To solve the problem, it is thus usually not sufficient to reduce the number of dock plants. The farm should be seen at the systems level. For example, it might be helpful to change animal housing on those farms using litter-minimised systems, to reduce the share of concentrate in the ration (and thus reduce the nitrogen available on-farm), seek an alternative to collecting animal manure as slurry, as well as seek a grassland management appropriate for the local ecological conditions (Dietl and Lehmann, 2004).

Another example is that nutrient cycles need to be considered when selecting the number and type of animals to be kept on a farm. A specialisation in monogastric livestock (pigs, poultry) frequently reduces the share of forage legumes in the crop rotation, because this type of livestock does not use plant material rich in fibre well. Besides other challenges, the long-term consequences of such a reduction may involve severe imbalances in the farm nutrient cycles and weed problems. Systems thinking and ensuring closed nutrient cycles is a foundation of organic farming and thus needs to be the basis of all farm decisions.

#### 5 Conclusion

Reports both in scientific journals and in popular media show that some of the developments within organic farming can lead to conventionalisation. By undermining the principles of organic farming, practices will undermine its transformative potential and thus its contribution to sustainable agriculture. However, we have shown that a number of the methods used in the conventionalisation debate do not reliably assess conventionalisation as they tend to focus on structural characteristics.

To be able to capture the changes at the farm level and be able to assess whether or not they amount to conventionalisation, a comprehensive framework built on the principles of organic farming would be useful. Indicators will need to capture the whole range of causes and effects as well as their interrelations and will clearly go above and beyond the requirements for organic certification. The indicators will need to comprehensively cover all four principles of organic farming defined by IFOAM (2005) and do justice to the systemic nature of organic farming. Defining criteria, deriving indicators and setting reference values need to be done in a participative process. The assessment framework could be used by, e.g., organic farmer associations to provide guidance and support extension activities. Farmer associations might implement a yearly assessment to encourage farmers to implement improved practices and thus to demonstrate their commitment to the organic principles. If the assessment includes a form of certificate, the farmers can use that certificate to convey their merits to customers.

Clearly, this would not prevent some farms from opting for conventionalised practices, but it would make their (lack of) commitment to organic principles transparent. This would allow addressing the increasing heterogeneity in organic farming practices. There are farmers practising 'organic light' and those that go 'beyond organic'. If the latter are to be supported and strengthened in their resistance to conventionalisation, their contribution to agricultural sustainability must be made visible. An assessment framework explicitly based on the principles of organic farming could contribute to both the reflexivity of organic farmers and to the academic discourse regarding the changes in and further development of organic farming.

Acknowledgements We would like to thank all those who participated in the conventionalisation workshop in May 2007 in Vienna for their open and constructive contribution and for their assessment of the changes in organic farming in Austria. We would like to thank AERU at Lincoln University for providing the supportive environment that made this paper possible. We are grateful to the anonymous reviewers and to the editor for their constructive and helpful suggestions on an earlier version of this paper. Of course, the authors alone are responsible for the content of the paper.

#### References

- Aeberhard A., Rist S. (2009) Transdisciplinary co-production of knowledge in the development of organic agriculture in Switzerland, Ecol. Econ. 68, 1171–1181.
- Allen P., Kovach M. (2000) The capitalistic composition of organic: The potential of markets in fulfilling the promise of organic agriculture, Agr. Hum. Values 17, 221–232.
- Alrøe H., Noe E. (2008) What makes organic agriculture move: protest, meaning or market? A polyocular approach to the dynamics and governance of organic agriculture, Int. J. Agr. Resour., Gov. Ecol. 7, 5–22.
- Andreoli M., Tellarini V. (2000) Farm sustainability evaluation: Methodology and practice, Agr. Ecosyst. Environ. 77, 43–52.
- Bähr M., Botschen M., Laberenz H., Naspetti S., Thelen E., Zanoli R. (2004) The European Consumer and Organic Food. Organic Marketing Initiatives and Rural Development, Vol. 4, University of Wales, Aberystwyth.

- Baars E., Baars T. (2007) Towards a philosophical underpinning of the holistic concept of integrity of organisms, NJAS-Wageningen J. Life Sci. 54, 463–477.
- Besson Y. (2008) Une histoire d'exigences : philosophie et agrobiologie. Paper presented at the conference "Développement et innovation en agriculture biologique" (DinABio) held 19–20 May 2008 in Montpellier, France.
- Best H. (2008) Organic agriculture and the conventionalization hypothesis: A case study from West Germany, Agr. Human Values 25, 95–106.
- BMLFUW (2007) Grüner Bericht. Bericht über die Lage der österreichischen Landwirtschaft 2007 [Report on the status of agriculture in Austria in 2007], BMLFUW (Federal Ministry of Agriculture, Forestry, Environment and Water Management), Vienna.
- Bowler I. (1992) The industrialisation of agriculture, in: Bowler I. (Ed.), The geography of agriculture in developed market economies, Longman, Harlow, pp. 7–31.
- Braband D., Geier U., Köpke U. (2003) Bio-resource evaluation within agri-environmental assessment tools in different European countries, Agr. Ecosyst. Environ. 98, 423–434.
- Brand K.W. (Ed.) (2006) Von der Agrarwende zur Konsumwende? Die Kettenperspektive [From the agriculturalturn to the consumption turn? The chain perspective]. Ergebnisband 2. Ergebnisse Sozial-Ökologischer Forschung Band 5. Oekom Verlag.
- Buck D., Getz C., Guthman J. (1997) From farm to table: The organic vegetable commodity chain of northern California, Sociol. Rural. 37, 3–20.
- Burch D., Lawrence G. (2005) Supermarket own brands, supply chains and the transformation of the agri-food system, Int. J. Sociol. Agr. Food 13, 1–18.
- CEC (2004) European action plan for organic food and farming, Communication from the Commission to the Council and the European Parliament, COM(2004)415 final, Commission of the European Communities, Brussels, 7 p.
- Coombes B., Campbell H. (1998) Dependent reproduction of alternative modes of agriculture: Organic farming in New Zealand, Sociol. Rural. 38, 127–145.
- DARD (2008) The code of good agricultural practice for the prevention of pollution of water, air and soil (COGAP), UK Department of Agriculture and Rural Development. Available on-line at: http://www.ruralni.gov.uk.
- Darnhofer I. (2005) Organic farming and rural development: Some evidence from Austria, Sociol. Rural. 45, 308–323.
- Darnhofer I. (2006) Organic farming between professionalisation and conventionalisation - The need for a more discerning view of farmer practices. Paper presented at Joint Organic Congress, Odense, Denmark, May 30–31, 2006. Available on-line at: http://orgprints.org/7390/.
- De Wit J., Verhoog H. (2007) Organic values and the conventionalization of organic agriculture, NJAS-Wageningen J. Life Sci. 54, 449–462.
- DeLind L.B. (2000) Transforming organic agriculture into industrial organic products: Reconsidering national organic standards, Hum. Organ. 59, 198–208.
- Degré A., Debouche C., Verhève D. (2007) Conventional versus alternative pig production assessed by multicriteria decision analysis, Agron Sustain. Dev. 27, 185–195.
- Dietl W., Lehmann J. (2004) Ökologischer Wiesenbau Nachhaltige Bewirtschaftung von Wiesen und Weiden

[Ecological grassland management – Sustainable management of meadows and pastures], Österreichischer Agrarverlag, Leopoldsdorf.

- DuPuis E.M., Gillon S. (2009) Alternative modes of governance: organic as civic engagement, Agr. Hum. Values 26, 43–56.
- EC (2004) European Action Plan for Organic Food and Farming. Commission Staff Working Document SEC(2004)739. Annex to the Communication from the Commission COM(2004)415final, 33 p. Available on-line at: http://ec.europa.eu/agriculture/organic/.
- EC (2007) Council Regulation (EC) No. 834/2007 of 28 June 2007 on organic production and labelling of organic products and repealing Regulation (EEC) No. 2092/91. Published in the Official Journal of the European Union L 189, on 20.7.2007.
- EEC (2001) Council Regulation (EEC) No. 2092/91 of 24 June 1991 on organic production of agricultural products and indications referring thereto on agricultural products and foodstuffs. Published in the Official Journal L 198, on 22.7.1991.
- Eurostat (2001) Organic farming. Statistics in Focus, Environment and Energy, Theme 8 – 5/2001, EuroStat, Luxembourg.
- Eurostat (2007) Different organic farming patterns within EU-25. An overview of the current situation. Statistics in Focus, Agriculture and Fisheries, 69/2007, EuroStat, Luxembourg.
- Fairweather J. (1999) Understanding how farmers choose between organic and conventional production: Results from New Zealand and policy implications, Agr. Hum. Values 16, 51–63.
- FiBL (2006) Ampferregulierung. Vorbeugende Möglichkeiten ausschöpfen [Dock regulation – Using preventive methods]. Merkblatt von Bioland, KÖN, Bio Austria und FiBL, 16 p. Available at http://www.fibl.org.
- Flaten, O., Lien G., Ebbesvik M., Koesling M., Valle P.S. (2006) Do the new organic producers differ from the 'old guard'? Empirical results from Norwegian dairy farming, Renew. Agr. Food Syst. 21, 174–182.
- Follett J. (2009) Choosing a food future: Differentiating among alternative food options, J. Agr. Environ. Ethics 22, 31–51.
- Frederiksen P., Langer V. (2008) Patterns of resource use on Danish organic farms: Aspects of farmbased rural development, Int. J. Agr. Resour., Gov. Ecol. 7, 96–109.
- Goodman D. (1999) Agro-food studies in the 'age of ecology': Nature, corporeality, bio-politics, Sociol. Rural. 39, 17–38.
- Green K., Foster C. (2005) Give peas a chance: Transformations in food consumption and production systems, Technol. Forecasting Social Change 72, 663–679.
- Guthman J. (1998) Regulating meaning, appropriating nature: The codification of California organic agriculture, Antipode 30, 135–154.
- Guthman J. (2000) Raising organic: An agro-ecological assessment of grower practices in California, Agr. Hum. Values 17, 257–260.
- Guthman J. (2004a) Agrarian dreams. The paradox of organic farming in California, University of California Press, Berkeley.
- Guthman J. (2004b) The trouble with 'organic lite' in California: A rejoinder to the 'conventionalisation' debate, Sociol. Rural. 44, 301–316.
- Guptill A. (2009) Exploring the conventionalization of organic dairy: Trends and counter-trends in upstate New York, Agr. Hum. Values 26, 29–42.

- Hall A., Mogyorody V. (2001) Organic farmers in Ontario: An examination of the conventionalization argument, Sociol. Rural. 41, 399–422.
- Häring A.M., Dabbert S., Aurbacher J., Bichler B., Eichert C., Gambelli D., Lampkin N., Offermann F., Olmos S., Tuson J., Zanoli R. (2004) Organic farming and measures of European agricultural policy. Organic Farming in Europe: Economics and Policy, Vol. 11, University of Hohenheim, Department of Farm Economics, Hohenheim.
- Hazel L.N. (1943) The genetic basis of constructing selection indexes, Genetics 28, 476–490.
- Hendrickson M., James H. (2005) The ethics of constrained choice: How the industrialization of agriculture impacts farming and farmer behavior, J. Agr. Environ. Ethics 18, 269–291.
- Holt G. (2006) A conceptual model of willingness to pay for organic food in the UK, in: Holt G., Reed M. (Eds.), Sociological perspectives of organic agriculture: From pioneer to policy, CABI, Wallingford, pp. 88–106.
- IFOAM (2005) Principles of organic agriculture, 4 p., IFOAM International Federation of Organic Agriculture Movements, Bonn, available on-line at: http://www.ifoam.org/.
- Jordan S., Shuji H., Izawa R. (2006) Conventionalization in the Australian organic industry: A case study of the Darling Downs region. In Sociological perspectives of organic agriculture: From pioneer to policy, CABI, Wallingford, pp. 142–156.
- Kaut A., Mason H., Navabi A., O'Donovan J., Spaner D. (2008) Organic and conventional management of mixtures of wheat and spring cereals, Agron. Sustain. Dev. 28, 363–371.
- Knickel K. (2008) Evaluating the environmental performance of farms in regionally adaptive and participatory ways, Paper presented at the conference on "Using evaluation to enhance the rural development value of agri-environmental measures" held 17–19 June 2008 in Pärnu, Estonia.
- Konefal J., Mascarenhas M., Hatanaka M. (2005) Governance in the global agro-food system: Backlighting the role of transnational supermarket chains, Agr. Hum. Values 22, 291–302.
- Lamine C., Bellon S. (2009) Conversion to organic farming: A multidimensional research object at the crossroads of agricultural and social sciences. A review, Agron. Sustain. Dev. 29, 97–112.
- Langer V. (2002) Changes in farm level structure following conversion to organic farming in Denmark, Ame. J. Alternat. Agr. 17, 75–82.
- Langer V., Frederiksen P. (2005) The development of farm size on Danish organic farms - A comment to the the conventionalisation debate. Presented at the ISOFAR, 21–23 September 2005 in Adelaide, available on-line at: http://orgprints.org/ 4406/.
- Langmeier M., Frossard E., Kreuzer M., M\u00e4der P., Dubois D., Oberson A. (2002) Nitrogen fertilizer value of cattle manure applied on soils originating from organic and conventional farming systems, Agronomie 22, 789–800.
- Letourneau D., Bothwell S. (2008) Comparison of organic and conventional farms: challenging ecologists to make biodiversity functional, Front. Ecol. Environ. 6, 430–438.
- Lobley M., Butler A., Reed M. (2009) The contribution of organic farming to rural development: An exploration of the

socio-economic linkages of organic and non-organic farms in England, Land Use Pol. 26, 723–735.

- Lockie S., Lyons K., Lawrence G., Mummery K. (2002) Eating "green": Motivations behind organic food consumption in Australia, Sociol. Rural. 42, 23–40.
- Lockie S., Halpin D. (2005) The 'conventionalisation' thesis reconsidered: Structural and ideological transformation of Australian organic agriculture, Sociol. Rural. 45, 284–307.
- Lockie S., Lyons K., Lawrence G., Halpin D. (2006) Going organic: mobilizing networks for environmentally responsible food production, CABI, Wallingford.
- Lotter D.W. (2003) Organic agriculture, J. Sustain. Agr. 21, 59–128.
- Lund V., Röcklingsberg H. (2001) Outlining a concept of animal welfare for organic farming systems, J. Agr. Environ. Ethics 14, 391–424.
- Luttikholt L.W. (2007) Principles of organic agriculture as formulated by the International Federation of Organic Agriculture Movements, NJAS-Wageningen J. Life Sci. 54, 347–360.
- Mäder P., Fließbach A., Dubois D., Gunst L., Fried P., Niggli U. (2002) Soil fertility and biodiversity in organic farming. Science 296, pp. 1694–1697.
- Malkina-Pykh I., Pykh Y. (2008) Quality-of-life indicators at different scales: Theoretical background, Ecol. Ind. 8, 854–862.
- Meul M., van Passel S., Nevens F., Dessein J., Rogge E., Mulier A., van Hauwermeiren A. (2008) MOTIFS: A monitoring tool for integrated farm sustainability, Agron. Sustain. Dev. 28, 321–332.
- Michelsen J., Hamm U., Wynen E., Roth E. (1999) The European market for organic products: Growth and development. Organic Farming in Europe: Economics and Policy, Vol. 7., University of Hohenheim, Department of Farm Economics, Hohenheim.
- Michelsen J. (2001) Recent development and political acceptance of organic farming in Europe, Sociol. Rural. 41, 3–20.
- Milestad R., Wivstad M., Lund V., Geber U. (2008) Goals and standards in Swedish organic farming: trading off between desirables, Int. J. Agr. Resour., Gov. Ecol. 7, 23–39.
- Moore O. (2008) How embedded are organic fresh fruit and vegetables at Irish farmers' markets and what does the answer say about the organic movement? An exploration using three models, Int. J. Agr. Resour., Gov. Ecol. 7, 144–157.
- Nieberg H., Offermann F., Zander K. (2007) Organic farms in a changing policy environment: Impacts of support payments, EU-enlargement and Luxembourg reforms. Organic Farming in Europe: Economics and Policy, Vol. 13, University of Hohenheim, Department of Farm Economics, Hohenheim.
- Niemeijer D., de Groot R.S. (2008) A conceptual framework for selecting environmental indicator sets, Ecol. Indicators 8, 14–25.
- Niggli U. (2007) Mythos "Bio". Kommentare zum gleichnamigen Artikel von Michael Miersch in der Wochenzeitung "Die Weltwoche" vom 20. September 2007. Commentary, FiBL, Frick, available on-line at: http://orgprints.org/11368/.
- Norton L., Johnson P., Joys A., Stuart R., Chamberlain D., Feber R., Firbank L., Manley W., Wolfe M., Hart B., Mathews F., Macdonald D., Fuller R. (2009) Consequences of organic and non-organic farming practices for field, farm and landscape complexity, Agr. Ecosyst. Environ. 129, 221–227.

- Obach B. (2007) Theoretical interpretation of the growth in organic agriculture: Agricultural modernization or an organic treadmill? Soc. Nat. Res. 20, 229–244.
- Padel S. (2007) How do ethical values of organic agriculture relate to standards and to current practice? in: Zollitsch W., Winckler C., Waiblinger S., Haslberger A. (Eds.), Sustainable food production and ethics, Wageningen University Press, Vienna, pp. 26–30.
- Padel S. (2008) Values of organic producers converting at different times: Results of a focus group study in five European countries, Int. J. Agr. Resour., Gov. Ecol. 7, 63–77.
- Padel S., Foster C. (2005) Exploring the gap between attitudes and behaviour – Understanding why consumers buy or do not buy organic food, Br. Food J. 107, 606–625.
- Padel S., Roecklingsberg H., Verhoog H., Alroe H., De Wit J., Kjeldsen C., Schmid O. (2007) Balancing and integrating basic values in the development of organic regulations and standards: Proposal for a procedure using case studies of conflicting areas, University of Wales, Aberystwyth, available on-line at: http://orgprints.org/10940/.
- Renting H., Oostindie H., Laurent C., Brunori G., Barjolle D., Moxnes Jervell A., Granberg L., Heinonen M. (2008) Multifunctionality of agricultural activities, changing rural identities and new institutional arrangements, Int. J. Agr. Resour., Gov. Ecol. 7, 361–385.
- Rigby D., Cáceres D. (2001) Organic farming and the sustainability of agricultural systems, Agr. Syst. 68, 21–40.
- Rinnofner T., Friedel J.K., de Kruijff R., Pietsch G., Freyer B. (2008) Effect of catch crop on N dynamics and following crops in organic farming, Agron. Sustain. Dev. 28, 551–558.
- Rosin C., Campbell H. (2009) Beyond bifurcation: Examining the conventions of organic agriculture in New Zealand, J. Rural Stud. 25, 35–47.
- Rosset P., Altieri M.A. (1997) Agroecology versus input substitution: a fundamental contradiction of sustainable agriculture, Soc. Nat. Resour. 10, 283–295.
- Schermer M. (2006) Regional rural development: The formation of ecoregions in Austria, in: Holt G., Reed M. (Eds.), Sociological perspectives of organic agriculture, CABI, Wallingford, pp. 227–242.
- Slee B. (2007) Social indicators of multifunctional rural land use: The case of forestry in the UK, Agr. Ecosyst. Environ. 120, 31–40.
- Smith E., Marsden T. (2004) Exploring the 'limits to growth' in UK organics: Beyond the statistical image, J. Rural Stud. 20, 345–357.
- Stock P.V. (2007) 'Good farmers' as reflexive producers: An examination of family organic farmers in the US Midwest, Sociol. Rural. 47, 83–102.
- Sundrum A. (2005) Vom ökologischen Landbau zur ökologischen Landwirtschaft [From organic agriculture to organic farming], Ökologie Landbau 1/2005(133), 17–19.
- Sylvander B., Bellon S., Benoit M. (2006) Facing the organic reality: The diversity of development models and their consequences on research policies, in: Organic farming and European rural development 58–59. Odense, available online at: http://www.orgprints.org/8247/.
- Thomas F., Groß D. (2005) Von der Bewegung zur Branche. Der Ökolandbau und seine sozialen und regionalen Ziele -Eine Diskussion über Anspruch, Realität und Perspektiven

[From movement to trade sector], in: AgrarBündnis (Ed.), Landwirtschaft 2005, Der kritische Agrarbericht, pp. 61–70.

- Tomlinson I. (2008) Re-thinking the transformation of organics: The role of the UK government in shaping British organic food and farming, Sociol. Rural. 48, 133–151.
- Tovey H. (1997) Food, environmentalism and rural sociology: On the organic farming movement in Ireland, Sociol. Rural. 37, 21–37.
- Trewavas A. (2001) Urban myths of organic farming, Nature 410, 409–410.
- USDA (2007) Organic agriculture: US organic farm sector is diverse. Briefing Rooms, Economic Research Service, United States Department of Agriculture, on-line: http://www.ers. usda.gov/briefing/organic/Farmsector.htm.
- van Cauwenbergh N., Biala K., Bielders C., Brouckaert V., Franchois L., Garcia Cidad V., Hermy M., Mathijs E., Muys B., Reijnders R., Sauvenier X., Valckx J., Vanclooser M., van der Veken B., Wauters E., Peeters A. (2007) SAFE – A hierarchical framework for assessing the sustainability of agricultural systems, Agr. Ecosyst. Environ. 120, 229–242.

- van der Ploeg J.D. (1995) From structural development to structural involution: The impact of new development in Dutch agriculture, in: van der Ploeg J.D., van Dijk G. (Eds.), Beyond Modernization: The impact of endogenous rural development, van Gorcum, Assen, pp. 109–146.
- van Ittersum M.K., Ewert F., Heckelei T., Wery J., Alkan Olsson J., Andersen E., Bezlepkina I., Brouwer F., Donatelli M., Flichman G., Olsson L., Rizzoli A., Van der Wal T., Wien J.E., Wolf J. (2008) Integrated assessment of agricultural systems - A component-based framework for the European Union (SEAMLESS), Agr. Syst. 96, 150–165.
- Verhoog H., Matze M., Lammerts van Bueren E., Baars T. (2003) The role of the concept of natural (naturalness) in organic farming, J. Agr. Environ. Ethics 16, 29–49.
- Watson C., Walker R., Stockdale E. (2008) Research in organic production systems – past, present and future, J. Agr. Sci. 146, 1–19.
- Willer H., Youssefi-Menzler M., Sorensen N. (Eds.) (2008) The world of organic agriculture. Statistics and emerging trends 2008, IFOAM and Frick: FiBL, Bonn, available at: http:// www.fibl.org/.

## **Conservation Tillage in Turkish Dryland Research**

Muzaffer Avci

#### Contents

1	Introduction	352
2	Fall Tillage in Fallow	353
3	Spring Tillage as Primary Operation	354
4	Summer Tillage	356
5	Economic Considerations	356
6	No-Tillage	357
7	Challenges and Implications	359
Re	ferences	360

M. Avci (🖂)

Abstract Central Anatolian soils have high risk of erosion, degradation and intensive cultivation. They are in danger of exhausting their agricultural use unless conservation agricultural practices are adopted. Conservation agriculture is a key tool in sustainable production systems throughout the world and is developed around soil management technology that minimizes soil disturbance, maximizes soil cover and promotes crop diversity to offer benefits to farmers and to environment. It has been particularly effective at sustaining crop production in semi-arid rain-fed regions such as the Central Anatolian soils, where potential evaporation exceeds precipitation during most months of the year, dry farming is extensively practiced, water and wind erosion is common, and proper application of water- and soil-conserving tillage technology is critical. The area under plow expanded its limits as the number of tractors in agriculture dramatically increased during the 1960s. This is the starting point for inappropriate use of the agricultural land. The conservation agricultural technologies, therefore, are of utmost importance for the region. Common farmers' practices of the current fallow-wheat system in the central plateau of Turkey are incompatible with the conservation agriculture concept. The objective of this review is to re-evaluate the performances of the partial and full conservation tillage practices previously tried in the region. The review reached the following conclusions: (1) agreeing with the conservation principles, fall tillage as a primary operation in the fallow areas was found to be useless compared with leaving the land without tillage; (2) therefore, much research has focused on spring tillage as a primary operation and employed conventional, semi-conservative and conservative methods. Results showed that the conventional

Central Research Institute Field Crops (CRIFC), Agronomy Department, PO Box 226, Ankara, Turkey e-mail: muzafavci@yahoo.com
system, in addition to being ecologically unfriendly, is unprofitable as compared with other conservation practices regarding the updated cost analysis; (3) similarly, tillage depth in primary spring tillage was determined to be shallower than the depths currently practiced by farmers, in agreement with the conservation principles; (4) fallow tillage operations in summer to create dust mulch for eliminating soil moisture loss did not increase the crop yields and soil moisture as compared with chemical fallow; (5) no-till fallow was similar to the conventional clean fallow system in terms of moisture and yield levels. However, no-tillage resulted in 50% reduction in the cost of tillage besides its ecologically-friendly effects; (6) the existing dryland agricultural systems in the plateau should be transformed into or changed toward sustainable systems, although further research is required on residue and stubble management, and integrated weed control methods, planting equipments and methods on the fields with high amounts of residue.

Keywords Dryland farming • Summer fallow • Erosion • Stubble • Straw • Tillage • No-tillage • Weed control

### **1** Introduction

Conservation tillage is variously defined around the globe. In the USA, the definition includes a minimum of 30% soil cover after planting to reduce soil erosion by water, or where soil erosion by wind is the primary concern, a minimum of 1120 kg ha<sup>-1</sup> of flat, small grain residue equivalent on the soil surface throughout the critical wind erosion period (CTIC, 2004). However, precise definitions of conservation tillage can only be made in the context of the crop species and varieties, soil types and conditions, and climates (Carter, 1994). Conservation agriculture is developed around soil management technology that minimizes soil disturbance, maximizes the soil cover and promotes crop diversity to offer benefits to farmers and to the environment.

Common agricultural practices regarding the fallow-wheat system in Central Anatolia are incompatible with the conservation agriculture concept. Surface soil has no residue cover and loses its aggregation and becomes dustlike structure as a result of frequent tillage operations during the 16 months of the fallow period. In addition to fallow fields, the planted fields have no crop residue cover and are under high risk of wind and water erosion during the growing period of 10 months, but especially immediately before and after drilling. Measures to protect soils from wind and water losses are the prerequisite for the sustainable use of the croplands. In Turkey, as the importance of water erosion is well known, wind erosion has been regarded as unimportant and confined to special areas such as Karapınar-Konya and İncesu-Kayseri towns. For the last 5 years, we have observed severe wind erosion (especially during the Lodos - south wind) in Central Anatolia during the winter months. Roadsides and sheltered fields are covered with the sediments eroded from open fields (Fig. 1). However, evidence of erosion is removed by rain or snowfall and soils then bear a resemblance to those affected by successive freezing and thawing events (Fig. 2). Soykan (1962) reported that the soils blown by wind accumulated in sheltered places and sometimes heaped up to the windows of the houses in Central Anatolia. He also emphasized the fact that soil erosion was started even by the gentle winds in the case of soils with no or little residue on them and low organic matter. Mutaf (1970) indicated that abrasion and eradication of young plants by wind and covering of planted fields with the eroded soil were regarded as ordinary events in Central Anatolia. Christiansen Veniger (1973) provided some photos emphasizing the wind erosion problems in the central plateau. In order to reduce the hazard of severe wind erosion, many researchers recommended reconsideration of the summer (clean) fallow practices which make the soil more vulnerable to wind erosion by successive summer tillage operations and construction of wind breaks (Beşkök, 1957; Beşkök, 1966; Hans Johaim Spath, 1975; and Aydemir, 1975). Soil



1 + + + + + + + +

Fig. 1 Field covered by sediments eroded by south winds (Lodos) in Central Anatolia



Fig. 2 Post-winter scene of soils eroded by south winds (Lodos) in Central Anatolia, 2006

surface cover of living vegetation or crop residue is the simplest and surest way of reducing erosion. Smallstemmed residues such as wheat or barley create more friction with the wind than large-stemmed ones such as corn or sunflower stalks. Standing residue is more effective than lying residue to control wind erosion.

The crop residue cover reduces evaporation, and the standing stubble traps snow for extra soil moisture. At the same time, the crop residue reduces wind speed on the soil surface, and standing stubble anchors the soil, and traps the snow (MWPS-45, 1992).

Most soils require 30 percent ground cover to prevent wind erosion. For cereal crop residues, this is equivalent to about 900 to 1100 kg ha<sup>-1</sup> of residue. The best way to reduce water and wind erosion is to leave the crop residue on the soil surface. The amount of residue mulch reduces erosion to a different extent. Residue mulches of 1.24 and 2.47 t ha<sup>-1</sup> minimized the loss of runoff water and soils on slopes of different lengths and steepness (Kramer and Meyer, 1969). While 2.0 t ha<sup>-1</sup> reduced the loss by 80%, a mulch rate of only 0.5 t ha<sup>-1</sup> decreased the loss by 40% (Lattanzi et al., 1974). In another study, erosion of 0.56 t ha<sup>-1</sup> and 1.12 t ha<sup>-1</sup> mulch rates was less than 25% of erosion without mulch (Meyer et al., 1970).

In the central plateau, crop residue research was carried out on fallow fields by using treatments with and without stubble (burning). The effects of the treatments of stubble management on the fallow fields were evaluated on the next years' crop phase. Thus, wheat stubble left on the field caused 110 to 950 kg ha<sup>-1</sup> less soil and 3.5 to 20 mm less soil water losses during the fallow phase, and 17 to 25% more subsequent wheat yields than stubble burning (Ayday, 1980; Sayın, 1983).

Besides soil erosion control, the soil organic matter content is increased by leaving stubble on the soil surface. There is consensus that carbon sequestration potential is higher in humid temperate areas (0.1 to 0.5 t C/ha/yr) than in semi-arid and tropical areas (0.05 to 0.20 t C/ha/yr). Palm et al. (2000) measured carbon stocks, losses and rates of accumulation in Brazil, Cameroon and Indonesia. They concluded that carbon accumulation rates are much higher in aboveground biomass (at least 2 t C/ha/yr) than in soils (0.2-0.6 t C/ha/yr), and also indicate that tree-based agro-ecosystems, either plantation crops (e.g. oil palm, cacao and rubber agro-forests) or on smallholder farms, bring the greatest dividend, accumulating 3.0-9.3 t C/ha/yr (Sanchez et al., 1999; Sanchez and Jama, 2000). Rapid decomposition occurred when the stubbles were buried to 12 cm depth and almost all the stubbles were mineralized in 18 months of the fallow period. On the other hand, only 33% of stubbles were decomposed when they were kept on the soil surface for the same period (Özbek et al., 1976; Cogle et al., 1987). Thus, a substantial part (60-65%) of the stubbles accumulates on the soil each year. Our observations also showed that no debris from the stubbles of the previous harvest can be seen on the field at the planting time of wheat. This clearly indicates that all the residue which was buried by the moldboard plow is decomposed during the fallow season.

The purpose of this review is to re-evaluate the performances of the conservation tillage practices previously studied in dryland conditions in view of the current socio-economic circumstances; and thereby shed light on the current and future research agenda by presenting the pros and cons of the practices.

#### 2 Fall Tillage in Fallow

The earliest studies on soil tillage in the fallow period were performed by Kıraç (1937) during the 1930s. In his studies, tillage practices were evaluated in terms of soil moisture conservation and nitrate accumulation as well as grain yield of wheat. Comparing his findings with those found in dryland areas of the USA, Kıraç discussed the results of fall tillage practices more prudently than his successors and tested various implements (moldboard plow, disc harrow and nontillage) for the following expectations: (1) controlling the weeds and volunteers during the fall; (2) burying weed seeds and eliminating them after germination; (3)increasing the rainfall penetration into the soil; and (4)enrichment of soil organic matter. Unfortunately, the results were in contrast to his expectations and one to 15% yield increase was obtained with non-tillage in the fall. Further investigating fall tillage, Gerek (1968) compared tillage and non-tillage on wet (good tilth) and dry soils with the control (moldboard plow in March) treatment under slope and bottom soils. The 18 years of results showed that while there was not any yield difference among the treatments on dry soil, significant differences were detected on bottom soils in favor of tillage. The advantage of tillage on bottom soil can partly be attributed to successful weed control by tillage treatment because the weed problem is expected to be more severe on bottom soils than slope soils. Volunteer and weed seeds germinate easily in the wet soil until the winter months and then are eliminated by the fall tillage. Thus, the fall tillage operation can contribute much to the weed control in wet soils. In another study, tillage treatments consisting of different reduced tillage methods and the moldboard plow were compared in 16 locations in Central Anatolia. Fall, spring and summer tillage practices were evaluated separately. The results of the fall tillage, although the results of statistical analysis was not given, showed that the moldboard plow provided more average yield than a chisel, subsoiler and notillage. The moldboard plow was superior to the other methods in only single year of the research. Therefore, by referencing the previous research results, the authors concluded no difference among the tillage methods, and recommended non-tillage in the fall for the central plateau (Pehlivantürk et al., 1977). Doğan et al. (1977) tested various fall tillage methods in the plateau for three years. The results indicated that chisel plowing was superior to non-tillage and the subsoiler in terms of grain yield and moisture conservation. However, the yield difference was not found to be statistically significant. Therefore, they decided in favor of non-tillage. As a summary of fall tillage, it was concluded by all the research results that tillage practice in the fall did not make a difference. This conclusion supports the idea of conservation agriculture. Leaving standing stubble on the field by non-tillage keeps the snow in the fields, increases the infiltration capacity of

M. Avci

the soil and reduces erosion. These facts make it unnecessary to till the field in the fall. If an area has soils which propagate weed infestation, or an impermeable soil layer, chisel plowing can be an alternative to nontillage for those areas.

#### 3 Spring Tillage as Primary Operation

Because the fall tillage as a primary operation in the fallow phase was shown to be a useless operation as compared with non-tillage, tillage in spring became the main concern for many researchers. Kıraç (1937) conducted his experiments by using the moldboard plow as a primary tillage tool. He compared and wanted to replace this costly equipment with cheap disc harrowing in March and April. He found equal yields and equal soil moisture but different nitrate contents. The amount of nitrate was high in moldboard-plowed plots, because intensive cultivation encouraged microorganism activity which led to rapid and complete decomposition of the wheat stubble. In those years, the wooden plow was common in the Anatolian plateau. The demand to use the moldboard plow instead of the wooden plow (saban) was very high because of its extensive use and popularity in the developed countries. The people advocating the wooden plow were said to be "reactionary" to modern agriculture. Kıraç (1937) handled the issue in a scientific way and tested the moldboard plow, wooden plow and lister as primary tillage implements for three years (1932 to 1934). The lister was preferred because it does similar work to the wooden plow. The results indicated that the wooden plow, contrary to expectations, was not as "dangerous" as a primary tillage tool and produced grain yields as high as the moldboard plow. However, it was stated by the author that it should be converted into an efficient metal tool by agricultural engineers. Gerek (1968) continued Kıraç's experiments on sloping and bottom lands for 18 years and found 1.32, 0.94, 1.00 and 0.99 t ha<sup>-1</sup> average grain yields with March moldboard plowing as a control, the wooden plow, lister and cultivator, respectively, on sloping land. Great yield differences between the implements in some years (1939, 40, 44, 46 and 48) were not discussed by the author. This makes it difficult to comment on those yield differences. However, Kıraç (1937) observed that the lister and the wooden plow were characterized by tilling the soils shallowly,

capturing the severe rainfalls of spring months in the soil and preventing wind and water erosion. As a consequence, we can conclude that the wooden plow and lister can be considered as the right equipment to be used in conservation agriculture and can replace the moldboard plow. On the bottom land, significant differences were not detected between the control (moldboard plow) treatment (1.48 t  $ha^{-1}$ ) and the wooden plow (1.41 t  $ha^{-1}$ ). The lister and cultivator yielded 12 and 15% less than the control treatment, respectively. Doğan et al. (1977) followed the recommendation made by Kıraç (1937) and modified the wooden plow into a metal form (Anatolian saban) which had a steel frame with several shanks. They compared the moldboard plow, the reduced moldboard plow, the metallic saban, Russian-type plow and sweep plow for four years. The modified metallic (saban) plow provided the highest yield in two years out of four and the second highest in the remaining other two years. The sweep plow also provided satisfactory performances. They recommended the Anatolian saban or the reduced moldboard plow as the primary tillage equipment (Doğan and Küçükçakar, 1987). All the research results showed that equipment similar to the wooden plow will do a satisfactory job on the plateau. Mutaf (1970) tested the moldboard plow with a small cutting edge in front, moldboard plows in different forms, disc plow, field cultivator, and sweep (Graham) plow for three years at the Polatlı State Farm. The highest wheat yield was obtained with the moldboard plow with a small cutting edge in front. The lowest yields were obtained with cultivator and subsoiling implements. He suggested that the cutting edge plow buries the structureless topsoil deeply and brings well-structured deeper soil to the surface; weeds, pests and organic residue on the surface are also buried deep in the soil, and humus is formed. Contrary to Mutaf's (1970) perception, much of the literature agrees that inverted soils lose their organic matter and structure rapidly (Mannering et al., 1975; Hill, 1990; Lal et al., 1998; Reicosky et al., 1995; Allmaras et al., 2000; Wilhelm et al., 2004). Inverting of soil one in two years in fallow/wheat sequences only postpones the degradation process and does not eliminate or reverse it. As was stated by Reicosky et al. (1995): "it is practically impossible to increase soil organic matter where moldboard plowing is taking place". The present situation on the plateau also shows that our soils lose aggregation and become tiny particles due to low organic

matter content (Eyüpoğlu, 1999) and primarily excessive moldboard plow tillage. Moldboard plow tillage and different reduced tillage methods were compared in 16 locations in Central Anatolia in the 1971-1977 period (Pehlivantürk et al., 1977). The moldboard plow was recommended for soils with slopes up to 8%. It was suggested that areas in the slope ranges of 3 to 8% should be plowed perpendicular to the slope direction, and areas with more than 8% of slope be allocating as grazing land. Those recommendations deserve some criticism. First of all, this perception did not consider wind erosion, that is not directly related to slope. Secondly, tilling the fields in the central plateau perpendicular to the slope direction is not a practical issue because most of the fields have a long border parallel to the slope direction. The last criticism is that a considerable amount of arable land on the plateau has slopes of more than 8% and there are no rules or laws acting to restrict the current use of those areas. Thus, the recommended method (moldboard plowing) and current extensive use of it by the farmers is not the right solution and involves none of the conservation measures that would prevent increased water and wind erosion.

The depth of tillage is very important in terms of water and soil conservation. The tilled layer retains more rain water than the untilled layer because of increased soil porosity. On the other hand, it accelerates the mineralization process of soil organic matter through enhanced oxidation. Surface (shallow) tillage results in lower soil disturbance and lesser amount of residue incorporation than deep tillage; consequently it leaves more residues on the soil surface (Sprague, 1986). Because the main aim of conservation agriculture is to leave a sufficient amount of plant residue on the seedbed, whatever technology that achieves this can be welcomed. Thus, the moldboard plow operated shallowly during tillage can leave more residues on the soil surface as compared with deep and complete inversion of the soil. Tillage depths were evaluated by several researchers in primary tillage operations in the central plateau. Deep tillage (20 to 25 cm), control (15 cm) and shallow (10 cm) were compared in terms of grain yields of wheat (Kıraç, 1937). The results showed that there was no difference between the deep and the control treatments and only a slight (2.5%) difference between the shallow and the control. Gerek (1968) tested the same treatments on different dates and in different soil conditions (on bottom and sloping soils) for 18 years. He

concluded that shallow (10 cm) tillage operation in April was superior to the control and the deep treatments on both the slope and the bottom lands. He reported that shallow tillage did not completely invert the soil, resulting in partly buried stubble. He suggested that this greatly prevents wind and water erosion. He also emphasized the fuel economy of shallow tillage and prevention of loss of fertility caused by mixing topsoil with infertile parent material brought up by deep operations in the shallow profile soils common in the central plateau. Berkmen (1961) tested 10, 20 and 30 cm of primary tillage depths for 7 years and was in favor of depths less than 20 cm. The difference between 10 and 20 cm was 2.6% of grain yield on average. Unver (1978) investigated time, implement and depth of primary tillage operations on soil moisture. Shallow tillage (11 to 13 cm) with the reduced moldboard plow and the sweep plow resulted in higher moisture than the moldboard plow. However, in cases of deep tillage (18-20 cm) the reverse was true. All of the research results we have discussed clearly demonstrate that, contrary to the current common farmers' practices of deep moldboard plow tillage, shallow primary tillage with the sweep plow and reduced moldboard plow, or even with the normal moldboard plow, tends to conserve more moisture than deep moldboard plow tillage, thereby emphasizing the importance of conservation agriculture for the region (Kiraç, 1937; Gerek, 1961; Berkmen, 1961; Ünver, 1978).

#### 4 Summer Tillage

The hot and dry summer period follows intensive rainy spring months on the plateau. Reducing soil moisture loss to a minimum during this period is the most important objective of the summer tillage. Moisture conservation can be achieved by reducing evapo-transpiration during this period. For this purpose, dust mulch and weed control are considered as two important tools. The significance of a surface dust layer (dust mulch) in terms of moisture conservation has always been emphasized when dry farming is the concern. The value of dust mulch was evaluated for moisture conservation. Four years of results indicated that weeds were the only reason for the moisture loss in fallow areas. Dust mulch caused slightly more (1%) moisture in the upper 30-cm layer, which can play a significant role in nitrate accumulation and crop emergence. Unfortunately, this extra moisture was not coupled with yield increases (Kıraç, 1937). Similar results were found when dust mulch and chemical weed control (no-tillage) were compared in terms of wheat yield and soil moisture (Karaca et al., 1981). The effects and importance of summer tillage operations in terms of conservation agriculture depend on the previous primary tillage operation on the fallow fields with standing and/or lying stubble. Inversion of soil as a primary tillage on those fields buries all the stubble into the soil. The subsequent tillage (summer tillage) operations will be of little or no value due to the unavailability of the crop residue on the field. Summer tillage research focusing on clean fallow is obviously beyond the scope of this paper (Doğan et al., 1977; Ünver, 1978; Pala, 1982; Doğan and Küçükçakar, 1987; and Güler et al., 1989). However, it is remarkable to see that following the same primary tillage methods, chemical weed control outperformed the summer tillage alternatives in terms of yield and net benefits (Table 2).

#### 5 Economic Considerations

Much of the research on tillage systems was carried out during the 1970s. The price of petroleum was much cheaper, but herbicide prices were very high. Hence, the economic analysis carried out for the tillage systems should be updated. Different tillage systems comprise equipment for primary tillage and a sweep harrow combination and chemical spray for summer tillage; these were tested in five provinces of the Anatolian plateau over five years (Annual Reports of Soils and Crop Managements of Wheat and Training Project, 1972 to 1977). According to the economic analysis (partial budget) on the updated costs and prices, yield losses with sweep tillage and offset disc harrows as primary tillage and chemical sprays as followup operations were compensated for by their reduced costs and reached identical profitability with moldboard plow systems (Table 2). An interesting result was the difference between tillage and chemical treatments the moldboard plow methods. The greater economic benefit of chemical sprays was due to their lower cost (Table 2). The economic analysis of the trials comparing different conservation tillage methods

with the conventional method at a state farm (Altinova, Konya) for three years (Annual Reports of National Wheat Project (Agronomy) for 1971, 1972 and 1973) also showed similar results. Chemical fallow provided 130 to 260 USD  $ha^{-1}$  more profit than the method of conventional fallow. It should be noted that the herbicides used in chemical fallow were contact herbicides and could not control all weeds. Today, we have nonselective, systemic herbicides which control almost all types of weeds. Although we spend more money with the current moldboard plowing than the conservation fallow system, there is no guarantee of obtaining more return with the conventional fallow technology due to crop hazards, e.g., cold, frost and drought, which are becoming more frequent due to climate change. Therefore, Kıraç (1937) formulated a principle for the farming system on the plateau; which may be valid for all dryland production systems. "We need the low cost technologies which may be subject to partial yield loss more than the costly technologies which may have potential yield increase".

#### 6 No-Tillage

No-till is defined as planting crops in previous crop residue and unprepared soil by opening a hole, narrow slot, trench or band of the smallest width and depth needed to obtain proper coverage of the seed. This is a single technology to achieve different goals simultaneously; it sustains agricultural production, and protects the environment. It is the most developed conservation agriculture technology, which has many agro-ecological advantages: controlling water and wind erosion by leaving crop residue on the field surface, increasing organic matter, preventing stubble burning, common on the plateau, and reducing the cost of production and greenhouse gas release to the atmosphere by eliminating tillage. Central Anatolian soils oxidize to two meters, and are friable due to high calcium content and frequent freezing and thawing. Thereby, the physical properties (tilth) expected from tillage occur by themselves under natural processes (Kirac, 1937). So, we do not see substantial differences among the tillage methods affecting soil properties and crop yields. This implies that tillage can be easily avoided if stubble and weed problems can be overcome in our soils. No-till fallow (hand weed control) and till fallow plots plowed at various depths as primary tillage and follow-up monthly cultivations were compared for soil moisture contents at wheat planting time for three years (K1rac, 1937). No considerable differences were found between primary tillage methods including different tillage depths and no-tillage in terms of soil moisture. Stubble mulch on till or no-till plots increased the amount of moisture stored and surface moisture. In dry years, no-till provided higher wheat yields than tilled plots, and vice versa in wet years. Soil moisture at 30 cm is less in notill than tilled plots (K1raç, 1937). This research clearly shows that if the weeds are controlled, no-tillage can replace the conventional fallow system. If straw was spread on the no-till plots (straw mulch), more soil moisture would be accumulated than in the conventional system. The results of only one season (1935/36) in which the soil was not frozen and allowed rainwater penetration into deeper zones show the effect of surface conditions (Table 1). In a three-year experiment, Kıraç (1937) observed deeper penetration of precipitation into the soil profile with no-till and residuecovered plots than tilled plots. The soils of no-till plots froze later and to a shallower depth and kept water longer than tilled plots. They were more effective in terms of moisture storage than hand-weeded control plots and plowed stubble cover and showed the benefit of conservation agriculture. The no-till system will play a vital role in preventing common stubble burning, reducing tillage costs by 30% and reducing soil and water loss. The research at the Altınova State Farm (annual reports for 1971, 1972 and 1973) showed that chemical fallow was more profitable, with the net benefit amounting to 260 USD ha<sup>-1</sup> than tillage systems. The research conducted by the CRIFC produced infor-

 Table 1 Impact of various tillage and surface management techniques on stored available soil water (in 180-cm profile) at the end of the rainy period (mid-June, 1932) (Kıraç, 1937)

	E
Types of management	Soil moisture (mm)*
Stubble cover	159
Sand cover	153
No-tillage + hand weed control	146
Plow in March at 15 cm depth	139
No-tillage + no weed control	132
Dry plowing at 50 cm depth	127
Soil dust (road dust) cover	95

\*Total precipitation during the experimental period was 226.2 mm.

	Tillage metho	ds				
	Sweep	Sween	Offset disc $\pm 2$ passes	Offset disc	M Plow	
Yield, cost and benefits	+ 2 passes of sweep + harrow combination	+ 2 times herbicide sprays	+ 2 passes of sweep + harrow combination	+ 2 times herbicide sprays	+ 2 passes of sweep + harrow combination (Control)	M. Plow + 2 times herbicide sprays
Average yields, kg ha <sup>-1</sup>	2118.0	2065.0	2245.0	2101.0	2335.0	2281.0
Yield differences, kg ha <sup>-1</sup>	217.0	270.0	90.0	234.0	0.0	54.0
Costs, USD ha <sup>-1</sup>	250.0	233.3	291.7	275.0	333.3	283.3
Benefit, USD ha <sup>-1</sup>	794.3	774.4	841.9	787.9	875.6	855.4
Net Benefits (NB), USD ha <sup>-1</sup>	544.3	541.0	550.2	512.9	542.3	572.0
NB diff. from Control	-2.0	1.3	-7.9	29.4	0.0	-29.7

**Table 2** Effects of different fallow tillage methods (primary + summer tillage operations) on average yields, and economic parameters

\* Average of 16 experiments carried out in different provinces of the plateau.

\*\* One pass of m. plow, sweep and offset disc and chemical spray costs 166.6, 83.3, 125.0 and 75.0 USD ha<sup>-1</sup>, respectively.

mative results on what the type and the structure of planting machines should be in terms of suitability for the Anatolian region for the success of the no-till system. As a result, two types of drills were developed. Studies on conservation agriculture and no-till started in 1998 and are ongoing. The seven years of research showed that the grain yields of wheat were similar in no-till and till systems; however, the cost of no-till was 50% less (134.5 USD ha<sup>-1</sup>) than the cost of a tillage system in fallow/wheat rotation (Avc1, 2005 and unpublished data). Similar results were also attained with a legume/wheat system. The Dutch Ankara Embassy supported no-till activities for the transfer of this technology to the Central Anatolian farmers during the 2003-2007 period. We conducted on-farm field trials and field demonstrations and farmer training during this period. In a village (Çeltikli-Haymana), the benefit from no-till (chemical fallowing) was 37% more than the farmers' system (clean fallow). This went up to 53% with the use of the improved wheat cultivar Ikizce96 instead of an existing variety, Bezostaya1. In another test area (Temelli-Ankara), no-till vetch hay was 1240 kg ha<sup>-1</sup> and 23% more than tillage vetch hay production. At the Polatlı State farm vetch (hay) and lentil yields were 0.5 t  $ha^{-1}$  and 0.45 t  $ha^{-1}$  more than the yields obtained from tillage systems. Two farmers in Bala and in Yerköy with large areas of land were in favor of the no-till system when they saw the benefits in the extremely dry season of 2006/2007 (project reports of 2003, 2005 and 2007). They would like to extend the

cooperation with research people and planned to plant a considerable amount of land with no-till systems. Similar climates to that of the plateau where conservation agriculture is well documented are in southern Australia and in the Pacific Northwest (PNW) of the USA. It will be instructive to use the experience of those regions in conservation agriculture. In southern Australia, the main elements in the conservation agriculture systems of direct drilling and stubble retention have been widely promoted. The benefits of conservation agriculture systems, which are reductions in erosion, increase in soil fertility, improvement in soil surface physical characteristics, and reduction in soil loss and bulk density, are well recognized (Poole, 1987; Steed et al., 1994). Despite these benefits the adoption process has been considered slow although about 36% of farmers in southern Australia use no-till and 80% retain stubble to some extent. The main reason for slow adoption is the expectation that the yield of crops in these systems will not increase relative to conventional systems; however, results from field experiments show that yields of wheat with conservation agriculture may be either higher or lower than with conventional tillage, and that there is considerable year-to-year variation both within and between sites (Kirkegaard, 1995). Conservation tillage in the PNW, like southern Australia, has been greatly influenced by soil degradation (Sojka and Carter, 1994), but particularly by the high rates of erosion and the need for water retention in the drier areas. Freeze-thaw events

and the associated problems with soil water storage and erosion have also been drivers in the adoption of conservation agriculture practices in the PNW. Like southern Australia, the rates of adoption have been low. In 2002, just 27% of the cropland in the states of the PNW was managed with conservation agriculture, and only 7.5% was managed with no-till systems (CTIC, 2004). The constraints to adoption are similar to those mentioned for southern Australia, with the exception of concerns over inadequate seed zone water for crop establishment under conservation agriculture. In the PNW, early seeding, often ahead of the first substantial winter rains, is vital for early establishment of the winter wheat to ensure adequate winter survival. The experience of the PNW and southern Australia in conservation agriculture shows that the main driving force in the adoption of conservation and no-tillage technologies was to obtain more yield or more returns with conservation than with conventional technologies. We are at the beginning in the adoption of conservation agriculture for the Anatolian plateau. Farmers do not have sufficient information on the concept of conservation agriculture. Environmental indicators (erosion, flood) are not taken seriously by farmers. The growing concern regarding erosion problems amid intellectual circles and some NGOs such as the TEMA Foundation is focused mainly on the non-agricultural sites (rangeland and forest).

#### 7 Challenges and Implications

Efficient and sustainable agricultural production requires that we continue to strive for systems that are efficient in their sustainability. In semi-arid regions of our country, conservation agriculture should be considered the primary standard by which systems and practices have been compared. Reduced productivity of our soils due to soil erosion, frequent tillage, intensive crop production and residue removal can be eliminated by reduced tillage systems. For example, previous research on tillage unanimously emphasizes non-tillage in the fall. For spring (primary) tillage, the Anatolian saban and the reduced moldboard and chemical fallow (no-tillage) outperformed in most of the experiments. However, there is not enough information about the residue levels and surface conditions in most of the studies. Therefore, studies on the amount and the state of crop residue (standing or lying of straw or stubble), and on the soil, water and crop yields are required for future research. On the plateau, demand for stubble grazing has been reducing because intensive animal production is replacing extensive systems. On the other hand, most of the straw is left in the fields besides the standing stubble in high-yielding years. Consequently, we have a chance to use the crop residue to improve and protect our soils. Some of the research (Karaca, 1987; Güler et al., 1989) highlighted the grassy weed population increase with reduced tillage and its control by the moldboard plow. Problems with grassy weeds can be overcome when we understand the weed biology. Seeds of downy brome buried five cm or more in the soil cannot emerge and only 1% survived after two years (Wicks et al., 1971; Durutan, 1982). Some of the chemical herbicides, metribuzin and diclofob, were found effective in control of downy brome (Durutan, 1982). New herbicides, Attribut (a.i. propoxycarbazoneNa) and Monitor (a.i. sulfosulfuron.), to control grassy weeds involving downy brome have been produced and they are effective when following the manufacturer's instructions. IPM (integrated pest management) research, which takes into consideration crop rotations, herbicide application and agronomic practices, is needed for grassy weed control in the dryland plateau. The unavailability of good drills which can perform well in high residue conditions was the main obstacle to obtaining high yields with reduced tillage methods in the past. In order to facilitate drilling with the available seed drills, chemical fallow fields were tilled prior to seeding to prepare a good seedbed. The performances of the available drills in no-tillage agriculture are very critical to the adoption of no-tillage agriculture by farmers on the plateau. Most producers will want to use machinery that is already owned. Some small modifications to the current drills by attachment of cutting disks should be tried in the fall and the spring plantings. Because the soil is very soft and swollen, the spring planting seems more feasible than the fall planting and also important for the implementation of no-tillage. Purchase of additional equipment adds significant costs unless it can be charged against production costs over enough hectares and years. The practical benefit of slightly more moisture in the upper zone of clean fallow than reduced tillage at planting time was not studied in detail. The interpretation was about the potential benefit on the germination and emergence and seedling failure "ala

tav" after planting (Kıraç, 1937; Pala, 1982). However, the research on crop emergence and its pre-winter development is very limited in the region.

Acknowledgements I gratefully acknowledge all the people who conducted or participated in the previous research activities mentioned in this paper, and also respectfully remember those who have passed away.

#### References

- Allmaras R.R., Schomberg H.H., Douglas C.L., Dao T.H. Jr., Campbell C.A., Lafond G.P., Zentner R.P., Biederbeck V.O. (2000) Soil organic carbon sequestration potential of adopting conservation tillage in US croplands, J. Soil Water Conserv. 55, 365–373.
- Avcı M. (2005) Zero and minimum tillage as alternatives to conventional cultivation in dryland fallow/wheat and annual cropping systems in Central Anatolia, in: Pala M., Beukes D.J., Dimes J.P., Myers R.J.K. (Eds.), Management for improved water use efficiency in the dry areas of Africa and West Asia proceedings of a workshop organized by Optimising Soil Water Use Consortium (OSWU), 22–26 April 2002, Ankara, Turkey, Aleppo, Syria, ICARDA, and Patanchero, India, CRISAT, pp. 89–101.
- Ayday E. (1980) Eskişehir yöresinde anız yakmanın su erozyonuna etkileri. Eskişehir: Eskişehir Bölge Topraksu Araştırma Enstitüsü Müdürlüğü Yayınları, Genel yayın 154, Rapor yayın 113, pp. 1–40.
- Aydemir H. (1975) Bala koruyucu orman şeritlerinin mikroklima ve tarımsal ürün verimlerine etkisi, Ormancılık Araştırma Enstitüsü Teknik Bülten 68, Ankara, pp. 1–40.
- Berkmen N. (1961) Ankara Zirai Araştırma Enstitüsü Çalışmalarından: 1931–60, Tarım Bakanlığı Ankara Zirai Araştırma Enstitüsü sayı 4, pp. 1–68.
- Beşkök T. (1957) Koruyucu orman şeritleri ve Bala'da koruyucu orman şeritleri tesisi denemeleri, Ormancılık Araştırma Enstitüsü Teknik Bülten 7, Ankara, pp. 1–21.
- Beşkök T. (1966) Trakya Bölgesinde rüzgar zararlarını önlemek maksadıyla yapılacak ağaçlandırmalar hakkında ön etüt raporu, Ormancılık Araştırma Enstitüsü, Teknik Raporlar 9, Ankara, pp. 1–32.
- Carter M.R. (1994) Conservation tillage in temperate agroecosystems, Lewis, Boca Raton, p. 390.
- Christiansen-Veniger F. (1973) Türkiye'de Tarla Kültürünün Temelleri, Menteb Matbaası, İstanbul 2, Baskı, pp. 185–194.
- Cogle A.L., Strong W.M., Saffigna P.G., Ladd J.N., Amato M. (1987) Wheat straw decomposition in Subtropical Australia II, Effect of straw placement on decomposition and recovery of added 15N urea, Aust. J. Soil Res. 25, 481–490.
- CTIC (2004) National crop residue management survey: 2002 results, Conservation Technology Information Centre, West Lafayette, http://www.ctic.purdue.edu/CTIC/CRM.html.
- Doğan O., Cimilli B., Gücer C. (1977) Orta Anadolu iklim koşullarında nadas-buğday tarım yönteminin uygulandığı yörelerde en uygun sonbahar, ilkbahar ve yaz toprak işleme aletleri ile bu aletlerle uyuşabilen mibzer çeşitlerinin saptanması üzerinde bazı araştırmalar, Merkez Topraksu Araştırma Enstitüsü, Yayın 45, Araştırma Rapor 12, pp. 1–45.

- Doğan O., ve Küçükçakar N. (1987) Kuru tarım yönteminin uygulandığı yörelerde en uygun bazı agro-teknik önlemler, Türkiye Tahıl Sempozyumu Kitabı, 6–9 ekim 1987 Bursa TUBITAK-TOAG, Uludağ Üniversitesi, pp. 39–49.
- Durutan N. (1982) Orta Anadolu Buğday Ekim Alanlarında Brom (*Bromus tectorum* L.) 'un Yayılışı, Biyoekolojisi ve Mücadele Olanaklarının Araştırılması (B. tectorum, survey, bioecology, control measures), A.Ü. Fen Bil. Ens. Yay, No: Bitki Koruma 6 (PhD), pp. 89–105.
- Eyüpoğlu F. (1999) Türkiye Topraklarının Verimlilik Durumu, T.C. Başbakanlık Köy Hizmetleri Gn. Md., Toprak ve Gübre Araştırma Enstitüsü Yayınları, Gn. Yayın 220, Teknik Yayın T-67, Ankara, pp. 5–75.
- Gerek R. (1968) Dryfarming İstasyonu Tarafından Yapılmış Olan Nadas Hazırlama ve Toprak Verimliliği, Neşriyat 6, 1–40.
- Güler M., Durutan N., Karaca M., Avçin A., Avcı M. (1989) Increasing WUE through fallow soil management under Central Anatolian conditions, in: Harris H.C., Cooper P.J.M., Pala M. (Eds.), Soil and crop management for improved water use efficiency in rainfed areas, Proc. Int. Workshop, Ankara, Turkey, pp. 76–84.
- Hans Johaim Spath (1975) Soil erosion and soil moisture balance in dry steppe climates with cold winters (as illustrated by the example of Central Anatolia), Geo Abstracts A6, 75A/2064.
- Hill R.L. (1990) Long-term conventional and no-tillage effects on selected soil physical properties, Soil Sci. Soc. Am. J. 54, 161–166.
- Karaca M. (1981) Nadas Toprak Islemesi Derinlik ve Yontemlerinin Infiltrasyona Etkileri, Ankara, Turkiye: A.U. Z.F, Doktora Tezi, pp. 45–55.
- Karaca M., Güler M., Pala M., Durutan N. (1984) Orta Anadolu'da Nadas Alanlarında Uygun Toprak Hazırlığı, in: Kuru Tarım Bölgelerinde Nadas Alanlarından Yararlanma Sempozyumu, 28–30 eylül, 1981, Ankara: TUBITAK Yayınları 593, TOAG 119, pp. 113–122.
- Karaca M., Güler M., Pala M., Durutan N., Ünver I. (1987) Orta Anadolu Bölgesinde Nadas Toprak isleme yöntemlerinin buğday verimine etkileri, in: Yürür N. (Ed.), Türkiye Tahıl Sempozyumu 6–9 ekim 1987, Bursa, TMO matbaası, pp. 29–39.
- Kıraç N. (1937) Eskişehir'de Dryfarming Araştırmaları, Dryfarming Deneme İstasyonu, İstanbul Teknik Belleten 1, İstanbul Devlet Basımevi, pp. 10–60.
- Kirkegaard J.A. (1995) A review of trends in wheat yield responses to conservation cropping in Australia, Aust. J. Exp. Agric. 35, 835–848.
- Kramer L.A., Meyer L.D. (1969) Small amount of surface mulch reduce soil erosion and runoff velocity, ASAE Transactions 12, 638–645.
- Lal R., Kimble J.M., Follett R.F., Cole C.V. (1998) The potential of US croplands to sequester carbon and mitigate the greenhouse effect, Ann Arbor Press, Chelsea.
- Lattanzi A.R., Meyer L.D., Baumgardner M.F. (1974) Influences of Mulch Rate and Slope Steepness on Interrill Erosion, Soil Sci. Soc. Am. J. 38, 946–950.
- Mannering J.V., Giriffith D.R., Richey C.B. (1975) Tillage for moisture conservation, Am. Soc. Agr. Eng., Paper No. 75–2523.
- Meyer L.D., Wischmeier W.H., Foster G.R. (1970) Mulch rate required for erosion control on stepped slopes, Soil Sci. Soc. Am. Proc. 34, 928–931.

- Mutaf E. (1970) Buğday ziraatında toprak işleme ve problemleri. In: Türkiye'de buğday yetiştiriciliği ve problemleri sempozyumu, 28–30 nisan 1969, TUBİTAK. TOAG, sayı:8, pp. 1–150.
- MWPS-45 (Midwest Plan Service Committee) (1992) Conservation tillage systems and managements: crop residue management with no till, ridge-till, mulch till, 1st ed., Agricultural and Biosystems Engineering Dep. Iowa State Univ. Ames IA50011–3080.
- Özbek H., Dinç U., Güzel N., Kapur S. (1976) Çukurova Bölgesinde anız yakmanın toprağın fiziksel ve kimyasal özellikleri üzerine etkisi, TUBİTAK, TOAG No: 182 Ankara. pp. 3–5.
- Pala M. (1982) Yaz Toprak İşleme Derinliklerinin ve Kullanılan Araçların Toprakta Nem ve Sıcaklık Değişimlerine Etkileri Üzerinde Bir Araştırma. Ankara: Ankara Üniversitesi Ziraat Fakültesi, Doktora Tezi, pp. 13–71.
- Palm C.A., Noordwijk M. van, Woomer P.L., Alegre J.C., Castilla C.E., Cordeiro D.G., Hairiah K., Kotto-Same J., Moukam A., Parton W.J., Ricse A., Rodriguesl V. (2000) Carbon losses and sequestration potentials of alternatives to slash and burn agriculture, in: The biology and fertility tropical soils, TSBF, Nairobi.
- Pehlivantürk A., İzgin C.N., Güler M., Ünver İ., Pala M. (1977) Orta Anadolu 1970–1976 buğday toprak hazırlığı ve yetiştirme tekniği araştırmaları Orta Anadolu Bölge Zirai Araştırma Enstitüsü yayınları, Yayın No: 77–2, pp. 5–15.
- Poole M. (1987) Tillage practices for crop production in winter rainfall areas, in: Cornish P.S., Pratley J.E. (Eds.), Tillage New Directions in Australian Agriculture, Inkata Press, Melbourne, pp. 24–47.
- Reicosky D.C., Kemper W.D., Langdale G.W., Douglas C.W., Trinsoutrot I. Jr., Recous S., Mary B., Nicolardot B., Rasmussen P.E. (1995) Soil organic matter changes resulting from tillage and biomass production, J. Soil Water Conserv. 50, 253–261.

- Sanchez P.A., Jama B.A. (2000) Soil fertility replenishment takes off in east Southern Africa. Paper for International Symposium on Balanced Nutrient Management Systems for the Moist Savanna and Humid Forest Zones of Africa, Cotonou, Benin, October 9, 2000.
- Sanchez P.A., Buresh R.J., Leakey R.R.B. (1999) Trees, soils and food security, Philos. Trans. R. Soc. Lond. B 253, 949–961.
- Sayın S. (1983) Eskişehir yöresi kuru tarım koşullarında bazı tarımsal uygulamaların su erozyonuna etkileri, Tokat Bölge Topraksu Araştırma Enstitüsü Müdürlüğü Yayınları, Genel yayın No. 58, rapor serisi No. 36, Tokat, 1983, pp. 15–25.
- Sojka R.E., Carter D.L. (1994) Constraints on conservation tillage under dryland and irrigated agriculture in the United States Pacific Northwest, in: Carter M.R. (Ed.), Conservation Tillage in Temperate Agroecosystems, Lewis Publishers, Boca Raton, pp. 285–304.
- Soykan İ. (1962) Orta Anadolu'da Rüzgar erozyonu, Topraksu sayı:11, pp. 1–26.
- Sprague M.A. (1986) Overview in: Milton A., Sprague M.A., Glover B., Triplett B. (Eds.), No tillage and surface tillage agriculture, John Wiley and Sons, 232.
- Steed G.R., Ellington A., Pratley J.E. (1994) Conservation tillage in the south eastern Australian wheat-sheep belt, in: Carter M.R. (Ed.), Conservation Tillage in Temperate Agroecosystems, Lewis Publishers, Boca Raton, pp. 231–254.
- Ünver I. (1978) Nadas toprak işlemesinde zaman, derinlik ve yöntemlerin toprağın rutubet ve sıcaklık değişimlerine etkileri, A.Ü. Ziraat Fakültesi Toprak bölümü, Doktora Tezi.
- Wicks G.A., Burnside O.C., Fenster C.R. (1971) Influence of soil type and depth of planting on downy brome seeds, Weed Sci. 19, 82–86.
- Wilhelm W.W., Johnson J.M.F., Hatfield J.L., Voorhees W.B., Linden D.R. (2004) Crop and soil productivity response to corn residue removal: a literature review, Agron. J. 96, 1–17.

# Part IV Climate Change

# **Biofuels, Greenhouse Gases and Climate Change**

Cécile Bessou, Fabien Ferchaud, Benoît Gabrielle, and Bruno Mary

#### Contents

1	Intro	duction	366
2	Defin	itions	368
3	Tran	sportation Biofuels	371
	3.1	First Generation Biofuels	312
	3.2	Worldwide	275
	3.3	Towards 2nd and 3rd Generations of Biofuels	382
4	Politi	ical and Economic Frameworks	388
	4.1	Climate Change and Greenhouse Gas	
		Emission Trends	388
	4.2	Biofuel-Related Policies	391
	4.3	Economic Incentives	395
5	Biofu	els and Greenhouse Gases	401
	5.1	Assessing the Environmental Impacts	
		of Biofuels	401
	5.2	Focus on Greenhouse Gas Emissions from	
		Agriculture	411
	5.3	Biofuel Greenhouse Gas Balances	426
6	The C	Quantitative Potential of Biofuels	434
	6.1	Biomass Availability: Bottom-up Approach	434
	6.2	Focus on Europe	447
	6.3	Liquid Biofuel Potential: Top-Down Approach.	448
	6.4	Projected Worldwide Biofuel Production and	
		Consumption	450
	6.5	Impact of Biofuels on Agricultural Commodity	
		Prices	453
7	Conc	lusion	455
Re	ferenc	2es	459

Abstract Biofuels are fuels produced from biomass, mostly in liquid form, within a time frame sufficiently short to consider that their feedstock (biomass) can be renewed, contrarily to fossil fuels. This paper reviews the current and future biofuel technologies, and their development impacts (including on the climate) within given policy and economic frameworks. Current technologies make it possible to provide first generation biodiesel, ethanol or biogas to the transport sector to be blended with fossil fuels. Still under-development 2nd generation biofuels from lignocellulose should be available on the market by 2020. Research is active on the improvement of their conversion efficiency. A ten-fold increase compared with current cost-effective capacities would make them highly competitive. Within bioenergy policies, emphasis has been put on biofuels for transportation as this sector is fast-growing and represents a major source of anthropogenic greenhouse gas emissions. Compared with fossil fuels, biofuel combustion can emit less greenhouse gases throughout their life cycle, considering that part of the emitted CO<sub>2</sub> returns to the atmosphere where it was fixed from by photosynthesis in the first place. Life cycle assessment (LCA) is commonly used to assess the potential environmental impacts of biofuel chains, notably the impact on global warming. This tool, whose holistic nature is fundamental to avoid pollution trade-offs, is a standardised methodology that should make comparisons between biofuel and fossil fuel chains objective and thorough. However, it is a complex and time-consuming process, which requires lots of data, and whose methodology is still lacking harmonisation. Hence the life-cycle performances of biofuel chains vary widely in the literature. Furthermore,

C. Bessou (🖂)

INRA Environment and Agricultural Crop Research Unit, 78850 Thiverval-Grignon, France e-mail: cecile.bessou@cirad.fr

LCA is a site- and time-independent tool that cannot take into account the spatial and temporal dimensions of emissions, and can hardly serve as a decision-making tool either at local or regional levels. Focusing on greenhouse gases, emission factors used in LCAs give a rough estimate of the potential average emissions on a national level. However, they do not take into account the types of crop, soil or management practices, for instance. Modelling the impact of local factors on the determinism of greenhouse gas emissions can provide better estimates for LCA on the local level, which would be the relevant scale and degree of reliability for decision-making purposes. Nevertheless, a deeper understanding of the processes involved, most notably N<sub>2</sub>O emissions, is still needed to definitely improve the accuracy of LCA. Perennial crops are a promising option for biofuels, due to their rapid and efficient use of nitrogen, and their limited farming operations. However, the main overall limiting factor to biofuel development will ultimately be land availability. Given the available land areas, population growth rate and consumption behaviours, it would be possible to reach by 2030 a global 10% biofuel share in the transport sector, contributing to lower global greenhouse gas emissions by up to 1  $GtCO_{2eq}$  per year (IEA, 2006), provided that harmonised policies ensure that sustainability criteria for the production systems are respected worldwide. Furthermore, policies should also be more integrative across sectors, so that changes in energy efficiency, the automotive sector and global consumption patterns converge towards drastic reduction of the pressure on resources. Indeed, neither biofuels nor other energy source or carriers are likely to mitigate the impacts of anthropogenic pressure on resources in a range that would compensate for this pressure growth. Hence, the first step is to reduce this pressure by starting from the variable that drives it up, i.e. anthropic consumptions.

**Keywords** Biofuels•Energy crops•Perennials•LCA• Greenhouse gases • Climate change • Political and economic frameworks•Bioenergy potential•Land-use change • Nitrous oxide • Carbon dioxide • Agricultural practices

#### **1** Introduction

Until the middle of the 19th century, American citizens lit their houses with whale-oil lamps. In 1892, the first Rudolf Diesel motor ran on peanut oil. Liquid fuels can be easily stored and transported and offer, for a given volume, a better exchange surface for combustion compared with solid fuels. Oils, in particular, can deliver a high energy amount by volume unit. No wonder then that biofuels were the first candidates to supply the newly developing automotive industry. However, they were almost immediately overtaken by petroleum products that appeared to be an energy godsend, remaining very cheap for more than a century. However, today the Black Gold Age is coming to an end.

In 2005, the world total primary energy supply approximated 11 430 Mtoe  $yr^{-1}$  (479 EJ  $yr^{-1}$ ), compared with 6130 Mtoe  $yr^{-1}$  (257 EJ  $yr^{-1}$ ) in 1973 (IEA, 2007a). According to the FAO, the world population will grow from around 6.5 billion people today to 8.3 in 2030 (UN, 2006). World energy demand is expected to rise by some 60% by 2030. More than two-thirds of the growth in world energy use will come from the developing countries, where economic and population growths are highest (CEC, 2006a). Fossil fuels will continue to dominate energy supplies, meeting more than 80% of the projected increase in primary energy demand. Global oil reserves today exceed the cumulative projected production between now and 2030, but reserves will need to be "proved up" in order to avoid a peak in production before the end of the projection period. Effective exploitation capacity today is almost fully used, so growing demand for refined products can only be met with additional capacity (IEA, 2005). The exact cost of finding and exploiting new resources over the coming decades is uncertain, but will certainly be substantial. Financing the required investments in non-OECD countries is one of the biggest challenges posed by our energy-supply projections

<sup>&</sup>lt;sup>1</sup> Mtoe yr<sup>-1</sup>: million ton oil equivalent: IEA conversion factor used throughout the article: 1 Mtoe yr<sup>-1</sup> =  $4.1868 \times 10^4$  TJ. Selected units for the article are Joules; however, conversions are indicated in brackets when quoted figures are given in other units.

(IEA, 2005). As an example, Saudi Arabia, with 25% of the world's best proven reserves, is already investing US \$50 billion to increase its production capacity by 2 million barrels per day (Mb/d); the global worldwide current production averaging 86 Mb/d (ASPO, 2008).

According to some experts the peak oil will occur in 20 years, whereas others argue that the world is already at peak production. Although one might argue on the exact moment, it is generally accepted now that it will happen soon and that an energy transition is unavoidable (Van der Drift and Boerrigter, 2006). ASPO, The Association for the Study of Peak Oil, states that overall oil and gas production will be at their peak by 2010, and the conventional oil peak would already be overcome in most regions (ASPO, 2008). According to Andris Piebalgs, European Union Energy Commissioner, the oil crisis of the 1970s presented a discrepancy between oil supply and demand of only 5%, but in a post-peak oil scenario, the gap between supply capacity and demand could reach 20% within five years (as quoted by ASPO, 2008). In 2030, the European Union energy dependency on imports could account for 70% of its global energy needs. Today, this dependency is already around 50% and the energy demand has kept increasing by 1 to 2% each year since 1986 (EU DG-TREN, 2005). Cheap reserves will not be sufficient to fulfil the world's growing energy demand and fossil fuels have been shown to be the main anthropogenic cause of global warming. These increasing supply and environmental costs make petroleum no longer the only candidate as the universal energy source; other sources may now be competitive. However, there is no alternative energy godsend and, as industries have been relying on petroleum for too long, clean technologies are late. There is no other solution than diversifying the energy mix with a growing contribution of cleaner energy sources.

Current global energy supplies are dominated by fossil fuels (81% in 2005), with much smaller contributions from nuclear power (6.3%) and hydropower (2.2%). Bioenergy provides about 10% of the total energy supplies, making it by far the most important renewable energy source used; solar, wind and other renewable energy sources accounting for the last 0.5% (IEA, 2007a). On average, in the industrialised countries biomass contributes less than 10% to the total energy supplies, but in developing countries the proportion is as high as 20–30%. In a number of countries biomass supplies 50–90% of the total energy demand. A considerable part of this biomass use is, however, non-commercial, and relates to cooking and space heating, generally by the poorer part of the world's population (IEA Bioenergy, 2007). The contribution of bioenergy to the global supply mix has scarcely evolved since 1973, whereas other renewable energy sources have been consequently fostered and nuclear power widely developed (IEA, 2007a). Bioenergy could play a bigger role, especially in the industrial countries, which consume a lot of fossil energy and are therefore the main contributors to atmospheric pollution and global warming. According to the intergovernmental panel on climate change (IPCC), greenhouse gas emissions have already made the world 0.6°C warmer during the last three decades. The EU-25 and the four other largest emitters, the United States, China, Russia and India, contribute all together approximately 61% of global emissions. Energy-related emissions represent 60% of global emissions in CO<sub>2</sub> equivalent (Baumert et al., 2005).

Transport is a major energy consumer [27.6% of total final energy consumption worldwide (IEA, 2007a), 31% in the EU-27 (EU DG-TREN, 2007); two-thirds of the projected increase in oil demand will come from transport (IEA world outlook 2005)] and a large greenhouse gas emitter. In 2004, the transport sector produced 6.3  $GtCO_2$ , i.e. 23% of world energy-related CO<sub>2</sub> emissions (Ribeiro et al., 2007) or roughly 13.5% of global greenhouse gas emissions (Baumert et al., 2005). In the EU-27, this sector accounted for 22% of total greenhouse gas emissions in 2005 (EEA, 2008). Moreover, vehicle emissions are the single most rapidly growing source of CO<sub>2</sub> emissions. Achievement of a levelling off of vehicle emissions, given continuing growth in the number of vehicles on the road, requires both: (1) a substantial reduction in vehicle emissions during the next several years and (2) advances in technology in the longer term that fundamentally reduce CO<sub>2</sub> emissions, because energy will always be at a premium (Hansen, 2006). The automotive market is logically evolving towards electric motors, whose energy efficiency is roughly 7.5 times higher than that of internal combustion engines. The compacity and lightness of liquid fuels still enable fifty-fold higher energy storage than the best current batteries (Roby, 2006). Fuel cells may in the future replace these limited electro-chemical accumulators, but these are considered long-term technologies requiring significant research and development efforts.

Their deployment also hinges on changes in the market and consumption behaviours. Lastly, electricity or  $H_2$  are secondary energy carriers that need to be produced from primary energy sources, involving possibly high CO<sub>2</sub> emissions.

Biofuels can contribute to reducing the dependency on fossil fuels and lower greenhouse gas emissions from transport, provided that the savings of greenhouse gases through the use of bioenergy is not counteracted by an increase in the same emissions during the production and transformation of the biomass. Agriculture and land-use change already account for some 15% and 13% of global greenhouse gas emissions, respectively (Baumert et al., 2005; Smith et al., 2007; Houghton, 2008). Can biofuels finally be considered as an advantageous clean energy source? Here, we address this question by first reviewing the various biofuels, the state of the art of the technologies, and the current production and consumption rates. We then present the political and economic frameworks that aim at promoting the development of biofuels but still fail at convincing all stakeholders about biofuel sustainability. We finally address the issue of biofuel quality in terms of environmental impacts, with a special focus on greenhouse gas emissions and the potential of biofuels to contribute to climate change mitigation.

#### 2 Definitions

This section provides definitions of the following key concepts: bioenergy, renewable, biofuels and biorefinery.

Bioenergy is the chemical energy contained in organic materials that can be converted into direct, useful energy sources via biological, mechanical or thermochemical processes. The most common and ancestral bioenergy source is firewood, which nowadays still represents 15% of global energy consumption (ADEME, 2006), some 90% of the world's wood-fuel being produced and consumed in the developing countries (Parikka, 2004). The prefix "bio", from the Greek " $\beta i \sigma \varsigma$ " (meaning "life")<sup>2</sup>, refers here to the

origin of the energy converted through the metabolism of living organisms with, at the basis of the food chain, autotrophic organisms converting solar energy into chemical energy contained in the molecules they produce via photosynthesis<sup>3</sup>. The total sum of living organisms is called the biosphere in opposition to lithosphere, hydrosphere and atmosphere. It is also referred to as "biomass" or "biota" by biologists and ecologists. Biomass, in the energy sector, refers to biological material which can be used as fuel for transport, or an energy source to produce industrial or domestic heat and electricity (feedstock and conversion routes, Fig. 1). Bioenergy comes from biomass. In contrast, fossil energies are mineral resources, stocked in the lithosphere. The carbon fossil energy sources are the result of mineralisation transforming organic matter into mineral matter. This transformation takes millions of years, meaning that fossil resources are non-renewable on a human time scale.

"Renewable" does not mean "sustainable". Renewable resources consist of two main types of natural resources: flow resources and renewable stock resources. Flow resources, such as solar or wind energies, are non-limited resources despite intermittence. On the contrary, renewable stock resources, mainly biomass, are limited resources and their availability depends both on other primary natural resources (e.g. lands, water, ecosystems, etc.) and on natural regeneration/degeneration rates and/or anthropic production/consumption rates. The term "renewables" in the energy field encompasses all energies coming from renewable resources, e.g. photovoltaic energy, wind energy, bioenergy, etc. It is also referred to as RES, standing for renewable energy sources.

Considering biomass, "renewable" indicates that it will in theory stay available in an infinite time perspective as it can regenerate or be grown. However in practice, the renewal of biomass also depends on its management, which should ensure that primary resources are not overexploited or even depleted. If resource management is technically appropriate, environmentally non-degrading, socially favourable and economically viable, then the renewable resource will

<sup>&</sup>lt;sup>2</sup> Thus the prefix "bio" has nothing to do with the organic production label called "BIO" in France or Germany, for instance, which actually corresponds to specific management guidelines

for agricultural production that aim at minimising the harmful impacts on the environment.

<sup>&</sup>lt;sup>3</sup> Other autotrophic processes than photosynthesis exist when enlacing the whole biosphere, but they are less relevant in quantitative terms when focusing on bioenergy.



Fig. 1 Bioenergy feedstock and conversion routes, adapted from Plassat, 2005 and UNDP, 2000. HTU: hydrothermal upgrading

be exploited in a sustainable<sup>4</sup> way. The issue of sustainability being crucial in the field of bioenergy, the UN Executive Board for Clean Development Mechanisms released in December 2006 an official definition of "Renewable Biomass" including this sustainability dimension (UNFCCC, 2006). Among the five possible conditions where biomass can be defined as "renewable", the three that do not deal with residues or wastes have a first criterion mentioning that the land use shall not change except if land areas are reverted to forest. The second criterion implicitly linked to the first one states: "Sustainable management practices are

undertaken on these land areas to ensure in particular that the level of carbon stocks on these land areas does not systematically decrease over time". This is a key element when comparing the  $CO_2$  emissions from biofuels and fossil fuels. Indeed, the interesting fundamental carbon neutrality of combusted biomass is based on the fact that the emitted  $CO_2$  from the plant originates from the atmosphere where it eventually goes back to. If land conversion to biomass production implies additional  $CO_2$  emission through soil organic carbon losses, it may offset this carbon neutrality.

Highly dependent on the type of vegetation, the organic carbon stock is globally around 1.6 times higher in forest or permanent grassland soils than in crop soils (Antoni and Arrouays, 2007). Therefore, land-use

<sup>&</sup>lt;sup>4</sup> Based on the pillars of sustainability concept in the Brundland report, 1987.

change can lead to soil organic matter losses. Soil organic carbon content can also decrease in the long term as a consequence of the export of agricultural residues. Therefore, it is necessary to assess the net crop residue amount that would remain available for the bioenergy chain without degrading the soil quality in the long term (Saffih-Hdadi and Mary, 2008; Gabrielle and Gagnaire, 2008). Soil organic matter represents only a few percent of the total soil mass, but still constitutes a large organic carbon stock on a global scale, i.e. almost the same as the sum of the carbon stocks in the atmosphere and in the vegetation (Arrouays et al., 2002). Small but stable changes in this stock could critically impact the global carbon fluxes. Furthermore, soil organic matter plays a crucial role in soil quality. In an agricultural soil, whose main function is to provide nutrients and water to crops, soil organic matter permits the development of microorganisms decomposing organic matter into easy-absorbed mineral forms for the plant. It contributes to the soil cationic exchange capacity, which also influences the availability of essential minerals for the plant, and to the stability of the soil. Soil organic matter and forest biodiversity are among the precious resources whose conservation for future generations should not be jeopardised by land-use changes. For the producers, "Renewable Biomass" will no longer just mean "which can be grown" but also implies conditions for a sustainable production.

Biofuels are biomass materials directly used as solid fuel or converted into liquid or gaseous fuels that can be stored, so that the harnessed energy can be released through combustion when needed. This chemical reaction permits the release of the binding energy that holds electrons to a nucleus in the organic molecules, in order to produce work and heat. In a narrower sense, biofuels may be only perceived as liquid or gas transportation fuels derived from biomass. Many different biomass raw materials can be used to produce biofuels including energy crops, agricultural residues or forest products, for example.

Biomass gives way to a whole product chain in which residues or co-products can largely contribute to the environmental and economic optimisation of the whole biomass value chain. Besides bioenergy, biomass can serve as a source of biomaterial (building materials, papers, etc.) and chemical compounds (solvents, pharmaceutics, cosmetics, biodegradable plastics, etc.). This last field of activities based on biomolecules is called "green chemistry". The overall integrated biorefinery that aims at using all the biomass compounds within one refinery complex is summarised in Fig. 2. The power of the biorefinery concept is supported by economy of scale and by efficient use of all incoming bioresources. Using the petroleum industry as an illustrative example,  $\sim 5\%$ of the total petroleum output from a conventional refinery goes to chemical products; the rest is used for transportation fuels and energy. Most visions for integrated biorefinery do not expect this ratio to change (Ragauskas et al., 2006).

To conclude, among renewables, biomass gives way to diverse bioenergy chains. In comparison with all the other renewables, bioenergy firstly presents the



Fig. 2 The fully integrated agro-biofuel-biomaterial-biopower cycle for sustainable technologies (Ragauskas et al., 2006)

advantage that investments are generally lower. Furthermore, the diversity of raw materials and transformation processes offers a wide range of possibilities than can be adapted to different geographical locations, means and needs. Nevertheless, issues or challenges also arise when dealing with the development of bioenergy that will aim at finding the best cost/benefit equations, including externalities, depending on both the type and the amount of bioenergy produced. These equations appear to be especially difficult to solve in the case of transportation biofuels. Firstly, the reverse side of feedstock and technology multiplicity is that many options from worst to best biofuel chains exist, which finally creates an overall confusion and uncertainty about biofuels. This uncertainty is thereby hampering the development of production plants and new expensive technologies as it contributes to increasing investment risks. Secondly, whereas bioenergy chains, like the other renewables, were intrinsically thought to be advantageous first of all on a local scale, biofuels especially give way to international trade, that can raise the issue of externality displacement and imply the need for specific political and economic frameworks.

#### **3** Transportation Biofuels

Biofuels are nowadays commonly classed as 1st, 2nd or 3rd generation biofuels, as shown in Table 1. First generation biofuels refer to those already consid-

 Table 1 Biofuel generations (Van der Drift and Boerrigter, 2006)

ered as "traditional or conventional chains", whereas 2nd generation biofuels, requiring more complex and expensive processes, are not available yet on the market. The energy efficiency of a biofuel chain must be appraised considering two aspects, both dependent on feedstock type: the net energy yield per area unit and the energy cost for transformation processes. When considering plant biomass, the energy yield per hectare is a function of the type of plant, the climate, the soil properties and the crop management. C<sub>4</sub> plants, whose photosynthesis is more efficient, are especially energy cost-effective in humid tropical regions where water is not limiting, e.g. sugar cane in Brazil. On the other end of the spectrum, maize in the US necessitates considerable energy inputs.

There is among 1st generation biofuels no technological breakthrough that would lead to large differences in terms of energy efficiency. In temperate regions, oilseed crops typically generate lower yields per hectare than sugar or starch crops and are therefore more expensive to produce. But because oils seeds require less processing they still generally have positive global energy balances per unit of feedstock. Oilseed crops grown in tropical areas can thus be especially productive and competitive. Globally feedstock costs account for the majority of the eventual price of any 1st generation biofuel, while processing costs and a small proportion for transport represent most of the rest. For ethanol, feedstock comprises 50 to 70% of the production cost, while for biodiesel feedstock can be 60 to 80% of the production cost (Lang et al., 2001; Worldwatch Institute, 2007).

Biomass feedstock	1st generation biofuels	2nd generation biofuels	3rd generation biofuels
Vegetable oil	Pure Plant Oil (PPO, also called		
	Virgin Plant Oil: VPO, or		
	Straight Vegetable Oil:VGO)		
	Fatty Acid Methyl Ester: FAME		
	(e.g. Rape Seed Methyl Ester)		
Fermentable biomass	Biogas/Substitute Natural Gas		H <sub>2</sub>
Starch/sugar	Ethanol/Ethyl Tertiary Butyl		
	Ether (ETBE)		
Lignocellulose		Ethanol	
		Fischer-Tropsch (FT) diesel*	
		Dimethyl Ether (DME)*	$H_2^*$
		Methanol*	
		Mixed Alcohols (MA)*	
		Substitute Natural Gas: SNG	

Biofuels indicated with \* are produced with synthesis gas (syngas, mainly H<sub>2</sub> and CO) as intermediate.

The split between 1st and 2nd generation biofuels lies in the fact that the latter are produced from lignocellulose, meaning that all types of vegetation and all parts of the plant are possible feedstock, whereas 1st generation biofuels only up-value specific parts of a few suitable plants. Hence, 2nd generation biofuels yield higher energy amounts per hectare than energy crops with a proportionally small specific organ of interest (such as seeds) as no part of the plant is left over. They also encompass a wider range of possible feedstock. Third generation biofuels are the follow-up of 2nd generation biofuels, from the same raw material up to H<sub>2</sub> production, whose energetic costs remain out of reach.

#### 3.1 First Generation Biofuels

The production of 1st generation biofuels could rapidly be fostered as technologies ensue from the food industry. Pure plant oils or even cooking oils, also called yellow grease, can thus be directly used as fuel. However, complementary processes permit one to upgrade the biofuels in order to optimise their mixing with conventional fossil fuels without needing to adapt the motors.

#### 3.1.1 Biodiesel

In a broad sense, biodiesel refers to pure and processed plant oils or animal fats. These oils and fats contain a mixture of triglycerides, free fatty acids, phospholipids, sterols, water, odorants and other impurities. Biodiesels are nowadays produced from a large range of oilseed crops, mainly rapeseed or canola, soybean and sunflower, palm oil and *Jatropha curcas* in tropical climates (see picture below). Other potential plant oil feedstock includes mustard seed, linseed, castor oil, peanut, cottonseed, coconut, *Lesquerella* spp. and micro-algae. There are as many different biodiesels as different oil compositions. Oilseed species vary considerably in their oil saturation and fatty acid content, characteristics that significantly affect the properties of the biodiesel produced.



Jatropha curcas fruit, Belize, February 2003 bessou©INRA

The boiling and melting points of the fatty acids, methyl esters and glycerides increase with the number of carbon atoms in the carbon chain, but decrease with increasing numbers of double bonds (Ma and Hanna, 1999). Saturated fatty acids are more compactable, which enhances the oil energy density. But if they contain lots of saturated fatty acids, oils and fats are solid at room temperature and cannot be directly used as fuel in a diesel engine in their original form except in warm climates. The disadvantages of vegetable oils compared with petroleum diesel fuel are their higher viscosity, lower volatility and the reactivity of unsaturated hydrocarbon chains (Lang et al., 2001). Because of subsequent problems such as carbon deposits in the engine, engine durability and lubricating oil contamination, they must be chemically transformed to be compatible and used on a long term with existing engines (Ma and Hanna, 1999).

The most widespread biodiesels are methyl esters produced from plant oils combined with methanol through transesterification. The two other routes, microemulsion and pyrolysis, are not worth it; pyrolysis notably is expensive for modest throughputs and processing removes any environmental benefits of using a biofuel (Ma and Hanna, 1999). Transesterification is an alkali-catalysed reaction that requires 107.5 kg of methanol per ton of vegetable oil and results in the production of 1004.5 kg of methyl ester and 103 kg of glycerol (Graboski and McCormick, 1998). In this three-step reaction, triglycerides are converted into diglycerides, then monoglycerides and finally reduced to fatty acid esters, enhancing the viscosity of the final biodiesel. The viscosity of vegetable oils and that of their final esters is of the order of 10-20 times and twice that of diesel fuel, respectively (Lang et al., 2001). A pre-step and catalysis make it possible to deal with the impurities such as free fatty acids and water to improve the reaction kinetics (Ma and Hanna, 1999). Methanol is preferred over ethanol because of its physical and chemical properties as well as comparative low cost (Ma and Hanna, 1999; Lang et al., 2001), although it introduces a part of fossil fuel into the biodiesel. For different esters from the same vegetable oil, methyl esters also appeared to be the most volatile ones (Lang et al., 2001).

Biodiesel used as an additive to diesel fuel can improve its lubricity. This property is becoming increasingly valuable as recent legislation has mandated further regulation on the sulphur content of diesel fuels; these cleaner diesel fuels exhibiting reduced lubricity as compared with their high sulphur predecessors (Radich, 2004; Goodrum and Geller, 2005). Some fatty acids such as ricinoleic (castor oil) and lesquerolic acids (Lesquerella spp.) could be especially efficient in enhancing the lubricity of a diesel fuel to an acceptable level at concentrations as low as 0.25% (Goodrum and Geller, 2005). Blending biodiesel with diesel fuel can increase the neat cetane number. Cetane number increases with increasing length of both fatty acid chain and ester groups, while it is inversely related to the number of double bonds and as double bonds and carbonyl groups move toward the centre of the chain (Graboski and McCormick, 1998; Tripartite Task Force, 2007). Highly saturated oils, with a low number of double bonds, hence provide the fuel with superior oxidative stability and higher cetane number (Worldwatch Institute, 2007). The average cetane numbers are 50.9 for soy and 52.9 for rapeseed esters. For the other esters listed in Graboski and McCormick, 1998, including sunflower, cottonseed and palm oil, cetane numbers vary in the 48-60 range. In comparison, the cetane index for petroleum diesel ranges from 40 to 52 (Radich, 2004).

The co-products of the entire chain are the meal left in the seed after oil extraction, which is sold as animal feed, and the glycerine from glycerol recovery, used in cosmetics. However, the rapid expansion of biodiesel has already saturated the market of glycerine in Europe, undercutting its ability to reduce the biodiesel price, as it could offset 5% of the production cost (Worldwatch Institute, 2007).

#### 3.1.2 Ethanol

Ethanol, on the contrary to biodiesel, is a singlecompound biofuel whose final composition does not vary with the type of feedstock. The feedstock is sugar and starch crops, which are basically equally processed through pre-treatment, fermentation by yeasts and other microbes, and distillation. The main sugar crops are sugar cane and sugar beet. Sweet sorghum could also become an interesting ethanol feedstock as a multi-use crop, whose seeds are edible and whose stalk contains sugar. The main starch crops used nowadays are maize and wheat; also potatoes, cassava and sorghum grain to a lower extent. Sugar crops typically yield more ethanol per hectare with an overall better energy balance than starch crops because (1) sugar crops yield higher sugar amounts per hectare compared with starch crops; and (2) sugar can be directly fermented, whereas starch long polymers have to be hydrolysed before being fed to yeast for the ethylic fermentation.

"Wet-milling" and "dry-milling" are the two current common methods to treat the starchy crop parts at their entry in the process chain. In the wet-milling, grains are soaked and chemically sub-divided into rich starchy parts of primary interest (grain endosperms) and other parts that contain more protein and fibres and constitute diverse co-products (maize oil and syrup, gluten feed, germ meal, vitamins and amino acids). These co-products can contribute up to 25% of the process economy (Worldwatch Institute, 2007). The dry-milling method only consists of grinding the unprocessed heterogeneous seeds into granules. It is therefore less expensive but also leads to less diverse co-product production. The main co-product is the dried distillers grain (DDG) fed to animals that can digest high proportions of fibres, and contributes up to 20% of the process economy (Worldwatch Institute, 2007). In both "wet" and "dry" processes, the starch

is finally hydrolysed into sugar typically using a high-temperature enzyme process (Fulton et al., 2004; Cardona and Sanchez, 2007).  $CO_2$  from fermentation can also be sold as a co-product to beverage industries. Indeed, in conventional fermentation, approximately one-third of the carbon available in the sugar is lost as  $CO_2$  (Strege, 2007).

The fermented ethanol must be distilled until enough water is removed to make the final anhydrous ethanol suitable for blending with gasoline (<1% of water in temperate climates). Indeed, water in ethanol blended with gasoline makes the fuel more sensitive to frost and increases the risk of phase separation in both storage and vehicle fuel tanks, which can cause serious operating problems for the engines (Tripartite Task Force, 2007; Balat et al., 2008). To improve the ethanol quality as a blend in low percentage, ethanol (47% on a mass basis) can be converted into Ethyl-Tertio-Butyl-Ether (ETBE) by reaction with isobutylene (53%). An ETBE blend of 15% corresponds to a blend in volume of 7% ethanol.

Since they both contain oxygen, ethanol and biodiesel are better combustibles than the substituted fossil oils, reducing the emission of pollutants such as CO, hydrocarbons (HC), sulphur oxide<sup>5</sup> and particulates by up to half of these emissions, depending on the biofuel and the blend mix (Shahid, 2007; Luneau and Fayet, 2007; Murugesan et al., 2008). Exhausts from blends with vegetable oils also depend on the engine load (Murugesan et al., 2008). Conclusions are not univocal concerning NOx emissions<sup>6</sup>, but biofuels would tend to lead to slightly higher emissions (Graboski and McCormick, 1998; Radich, 2004), notably with blends of 20% of ethanol or biodiesel in a car driven in the city (Luneau and Fayet, 2007). In Murugesan et al. (2008), NOx emissions are reported to be in the range of  $\pm 10\%$  as compared with diesel, depending on engine combustion characteristics (2008). Biofuels increase the octane level<sup>7</sup> (thanks to ethanol, Harijan et al., 2007; Balat et al., 2008) and cetane number (thanks to biodiesel, Radich, 2004). On the other hand, both ethanol and biodiesel may cause corrosion and are sensitive to cold weather.

The primary asset of biofuels is the convenience that they can be used as blends with conventional fuels in existing vehicles. However, unmarked blends are limited to certain extents according to fuel and vehicle specifications. As an example, in Europe, these blends on a volume basis are: 5% ethanol or 15% ETBE blends with gasoline, and currently 5% biodiesel in diesel fuel (Wiesenthal et al., 2007), up to 20–30% for specific fleets (Plassat, 2005). Beyond these limits, engines have to be adapted so that their performances will not be affected in the long run. This is, for instance, the case with the flexible-fuel vehicles that can run on low- and high-level ethanol blends up to 85–100% (also written as E85, E90, E100; biodiesel blends are noted B20, B30).

Divergences in biofuel technical specifications have been introduced worldwide due to feedstock variances, climatic conditions in each country and region, and the characteristics of the local markets (Tripartite Task Force, 2007). In consequence, automotive sector and biofuel blend-related strategies have also diverged. Diverse biofuel and blend standards were adopted worldwide by the various agencies, ABNT/ANP<sup>8</sup> in Brazil, ASTM International in the US and CEN<sup>9</sup> in Europe. The Tripartite Task Force has been working on implementing a road map to come up with international compatible biofuel-related standards to help increase the use of biofuels and avoid adverse trade implications in the global market.

The water content of ethanol shows how critical it may be to agree on standardised blends. Phase separation due to water occurs more readily at lower levels of ethanol in gasoline. In Brazil and the US, where ethanol blends reach up to 5.7–10% and 20–25%, respectively, phase separation is not an issue, whereas it becomes an issue in Europe with lower ethanol blends in gasoline. Additional dehydration increases production cost and can reduce productivity at the mill by up to 7% (Tripartite Task Force, 2007). Considering the additional environmental and economic costs related to biofuel upgrading, international harmonisation should urge a maximum level of blend flexibility correlated with a focus on minimum processing and optimum engine modifications.

<sup>&</sup>lt;sup>5</sup> Sulphur oxides (SOx) contribute to acid rain and can be carcinogenic.

<sup>&</sup>lt;sup>6</sup> NOx are precursors to the formation of tropospheric ozone.

<sup>&</sup>lt;sup>7</sup> Gasoline high octane value indicates a smaller likelihood that the fuel combusts too soon (low auto-ignite tendency), provoking engine knock problems. A high tendency to auto-ignite, or low octane rating, is undesirable in a spark ignition engine (gasoline) but desirable in a diesel engine (high cetane number).

<sup>&</sup>lt;sup>8</sup> Brazilian Petroleum, Natural Gas and Biofuels Agency.

<sup>9</sup> Comité Européen de Normalisation.

#### 3.1.3 Biogas

Biogas is produced through methanisation, i.e. the anaerobic digestion by bacteria of biodegradable matter such as municipal solid or agricultural waste, liquid slurry, solid manure, or maize silage, for instance. The more dry matter and fatty acids in the substrate contents, the greater the biogas yield (Moras, 2007). Apart from about 55 to 70% of methane (52–65% in examples in Table 2), the actual fuel, biogas also contains substantial amounts of  $CO_2$ , 30 to 45%, small quantities of hydrogen sulphide and other trace gases such as ammonia<sup>10</sup>. The separation of these components of biogas via a gas scrubber is an expensive prerequisite in order to use the biogas as fuel or to mix it with natural gas.

Biogas is less considered as transportation biofuel, because its target vehicle fleet remains marginal due notably to onboard gas storage constraints. The primary interest of biogas remains its local development as fuel for heat and power plants in rural areas. About 25 million households worldwide currently receive energy for lighting and cooking from biogas produced in household-scale digesters, including 20 million in China, 3.9 million in India and 150 thousand in Nepal (REN21, 2008). Hence, the two prevalent types of digesters are the Chinese "fixed dome" and the Indian "floating cover" that only differ by the gas collection method (ITDG, 2000). Biogas production in specifically designed digesters is the most widespread technology, although capturing methane from municipal waste landfill sites has been considerably developed. In the US, waste management including the recovering of methane produced by landfills has made possible to reduce these methane emissions by 50% over the years and has become one of the largest holders of greenhouse gas emission credits (Kram, 2007). Although the reaction takes several days to finally degrade just about 10 to 15% of the initial material, biogas permits one to take advantage of cheap feedstock and to diminish greenhouse gas emissions. Moreover, the solid residue of the process can be dried and used as fertiliser that has a high nutrient content and whose pathogenic germs have been killed during the digestion process, notably due to temperatures of 35°C to more than 50°C (ITDG, 2000; Baserga, 2000).

<sup>10</sup> http://www.planete-energies.com (consulted on 10.03.2008).

375

Biogas as transportation fuel could receive more attention in the coming decades, especially for use in city fleets and trucks, as has been the case in Sweden and Switzerland for a long time. Table 2 gives the productivities of common feedstock and biofuels worldwide.

# 3.2 Current 1st Generation Biofuel Supply Worldwide

Combustible renewables and waste are mostly con sumed directly; only about 1.8% is consumed by the transport sector, about 17.6% by the industries and 80.6% by other sectors, notably households (IEA, 2007a). Production of heat and electricity dominate current bioenergy use with two key industrial sectors for application of state-of-the-art biomass combustion for power generation: the paper and pulp sector and cane-based sugar industry (IEA Bioenergy, 2007).

Global fuel ethanol production more than doubled between 2001 and 2006, while that of biodiesel expanded nearly six-fold. The US and Brazil currently dominate world ethanol production, which reached a record 52 billion litres in 2007 (FAO, 2008a). Close to half the world's fuel ethanol was produced in the US from 14% of its national maize production in 2006 (Möller et al., 2007), and more than two-fifths in Brazil from sugar cane, roughly providing 21%<sup>11</sup> of its transport fuel consumption (OECD, 2006). The remaining production comes from Spain, Sweden, France and Germany. China's ethanol from maize, wheat and sugar cane is mostly destined for industrial use.

In 2006, Europe accounted for 73% of all biodiesel<sup>12</sup> production worldwide, mainly from rapeseed and sunflower seeds, with Germany as the leading producer (40%), followed by the US, France and Italy generating most of the rest (Worldwatch Institute, 2007). In 2007, the EU still accounted for 60% of global biodiesel production, that amounted to 6.2 billion litres in 2007 (FAO, 2008a), but biodiesel production has increased in all producing countries; it has doubled in the USA and in most of the producing

<sup>&</sup>lt;sup>11</sup> Various data: 13.2% in energy terms according to IEA 2006; 40% according to Xavier (2007).

<sup>&</sup>lt;sup>12</sup> Biodiesel here does not take into consideration pure vegetable oils mostly directly consumed by farmers on the farm.

Table 2 M	ain feedstock,	productivity pe-	r hectare, and co-proc	ducts of 1st generation bio	ofuels; FM: fresh ma	atter	
Biomass	Countries	Yields		Fossil equivalent (on energy basis)	Total energy output/Fossil		
feedstock	regions	t FM ha <sup>-1</sup>	Biofuels L ha <sup>-1</sup>	$L ha^{-1}$	energy input	Co-products	References
SUGAR CROPS			ethanol $(11 = 0.791)$	1L  ethanol = 0.665 L	Indicative values	Crop residues left in the field are often considered as fertilizing	Fulton et al., 2004 FNR, 2007
			kg)			co-products and are important for sustaining soil C content	
SUGAR CANE	Brazil	85	7.080	4.708		Yeast as cattle feed supplement;	Brazilian Ministry in Xavier, 2007
			6.500	4.323		Bagasse as feedstock for feed heat. Electricity, and cellulosic ethanol	Cardona and Sanchez, 2007
			6.800	4.522		Fructose as sweetener; invertase for food industry 23.8 t ha <sup>-1</sup> bagasse (50% H <sub>2</sub> O) - 7 854 t Wh electricity	Fulton et al., 2004
		73.5	5.476	3.641		Energy from cane trash	Moreira, 2000
		60.7	4.522	3.007	~8 [7 1 8 2]	3	Naylor et al., 2007 in FAO,
					[C:0-1:7]		2008a Naylor et al., 2007 in FAO, 2008a
	India	66.5–71.3–	5.300	3.525		2.753-2.952 litres of cellulosic ethanol from $20.5-22$ t ha <sup>-1</sup>	Fulton et al., 2004 Kadam, 2000
	Thailand	55.7	I	I		bagasse 104 kg sugar/t cane +45 kg	Nguyen and Gheewala,
						motasses/t cane (+5 kg rice husk) = 10.2 L ethanol [national availability~1Mt	2008 Prasertsri, 2006
	Global	65	4.550	3.026		y1 III01455C5]	Rajagopal et al., 2007 in FAO, 2008a
SUGAR	France	66.2	3.793	2.522		$3.7 \text{ t ha}^{-1}$ of sugar	ADEME/DIREM, 2002
BEET		I	7.307 7.500	4.860 4.988		If only ethanol without sugar production 44 t green manure (10% dry matter)	

France Betteraves/Passion Céréales, 2007 FNR, 2007 Fulton et al., 2004 JRC/EUCAR/CONCAWE, 2008 Raiagobal et al., 2007 in	FAO, 2008a USDA, 2007	Brazilian Ministry in Xavier, 2007 Fulton et al., 2004 Cardona and Sanchez, 2007 Graboski, 2002 Naylor et al., 2007 in FAO, 2008a Levelton, 1999 USDA, 2007 Naylor et al., 2007 in (FAO, 2008a) France Betteraves/Passion Céréales, 2007 Rajagopal et al., 2007 in FAO, 2008a
(Stabilized) molasses as fertiliser Sugar beet pulp and dried slop as animal feed Slop to biogas Heat	3–4 t sorghum grains	Wet-milling process : fibre as substrate for enzymes production, CCDS as feed for non-ruminants, gluten meal (60% protein. high fat) and gluten feed (20% protein, low fat), corn oil, different chemicals and food-related products as vitamins and amino acids Dry-milling process: DDGS as feed for ruminants (27–35% protein) 4.8 t ha <sup>-1</sup> DDGS
~2 [1.2-2.2]	~1 [0.9–1.1]	~1.5 ~1.5 [1.4-1.8]
3.617 3.658 - 3.365	3.370 5.057	2.374 2.062 2.308 2.494 2.250 1.349 1.327 2.394 1.303
5.439 5.500 - 5.060	5.070 7.605	3.570 3.100 3.471 3.384 3.384 3.384 1.995 1.995 3.600 1.960
55.5 - 68.9 46	09 06	10 9.4 7.2 5 5 4.9 4.9
Germany EU Global	et China DRGHUM CH	ROPS EE Brazil USA Canada China France Global
	SWEF S( STAR	MAIZ

Table 2 (contin	(pənı						
				Fossil equivalent	Total energy		
Biomass feedstock	Countries regions	Yields t FM ha <sup>-1</sup>	Biofuels L ha <sup>-1</sup>	(on energy basis) L ha <sup>-1</sup>	output/Fossil energy input	Co-products	References
WHEAT	France	7.5 9	2.600 2.550	1.729 1.696		2.8 t ha <sup>-1</sup> DDGS DDGS can be used as animal feed or for fuel	FB/Passion Céréales, 2007 ADEME/DIREM, 2002
	Germany	I	2.560	1.702		Glycerol	FNR, 2007
		7.44	2.753	1.831	~2	Straw for fertiliser or fuel	Öko-Institut, 2004
	ĒŪ	- 5.2	2.500	1.663 -	[1.2-4.2]		Fulton et al., 2004 JRC/EUCAR/CONCAWE.
							2008
	Global	2.8	952	633			Rajagopal et al., 2007 in FAO, 2008a
CASSAVA	Brazil	13.6	1.863	1.239			Naylor et al., 2007 in FAO, 2008a
	Nigeria	10.8	1.480	984	$\sim 2.85$		
	Thailand Global	19 12	- 2.070	- 1.377	2.1 in China		Nguyen et al., 2007 Rajagopal et al., 2007 in
							FAO, 2008a
RICE	Global	4.2	1.806	1.200	I		Rajagopal et al., 2007 in FAO, 2008a
BARLEY	Global	I	1.100	732	I	DDGS can be used as animal feed or for fuel	Fulton et al., 2004
SORGHUM GRAIN	Global	1.3	494	329	I	Straw for fertiliser or fuel	Rajagopal et al., 2007 in FAO, 2008a
OIL PLANTS			Pure vegetable oils	rapeseed oil $1 L = 0.961$			ITDG, 2000
				aneser ruer sunflower oil			A DEME/DIREM 2002
				1 L = 0.851 diesel fuel			FNR, 2007
SUNFLOWER RAPESEED	France	3.34 2.44	1.522 1.178	1.461 1.000	4.7 7.5	Meal or cake as animal feed	ADEME/DIREM, 2002
	Germany	3.4	1.480	1.420	3-5		FNR, 2007

Worldwatch Institute, 2007 ITDG, 2000 Lang et al., 2001	FNR, 2007 ADEME/DIREM, 2002 Fulton et al., 2004 JRC/EUCAR/CONCAWE, 2008	ADEME/DIREM, 2002 Fulton et al., 2004	JRC/EUCAR/CONCAWE, 2008 Fulton et al., 2004	Naylor et al., 2007 in FAO, 2008a	Fulton et al., 2004 Fulton et al., 2004	Naylor et al., 2007 in FAO,	JRC/EUCAR/CONCAWE, 2008	(continued)
		Meal or cake as animal feed or converted to biogas, steam and aborticity	Glycerol sold or converted to biogas then steam					
95 L 1.85 L 0.89 L	~2.5 [1.3–3.7]	ε			~3 [1,5–3,5]			
1 L biodiesel = $0.88-0$ diesel = ( diesel = ( diesel = ( diesel = (	1.380 1.944 1.068 -	1.504	890	445 492	623 356	437	I	
Biodiesel (methyl-ester) $15^{\circ}C (1 L)$ = 0.9 kg)	1.550 2.184 1.200 -	1.690	1.000	500 552	700 400	491	I	
	3.4 3.34 - 3.11	2.44	- 2.44	- 2.7		2.4	2.8	
	Germany France EU	France	EU	USA	EU	Brazil		
METHYL- ESTERS	RAPESEED	SUNFLOWER		SOYBEAN				

Table 2 (contin	nued)						
i i		-		Fossil equivalent	Total energy		
Biomass feedstock	Countries regions	Yields t FM ha <sup>-1</sup>	Biofuels L ha <sup>-1</sup>	(on energy basis) L ha <sup>-1</sup>	output/Foss11 energy input	Co-products	References
PALM OIL	Brazil	I	5.000	4.450			Fulton et al., 2004
	Malaysia	I	6.000	5.340			Fulton et al., 2004
		20.6	4.736	4.215	6~	Glycerol sold or converted to	Naylor et al., 2007 in FAO,
					[8.7–9.7]	biogas then steam	2008a
	Indonesia	17.8	4.092	3.642			Naylor et al., 2007 in FAO, 2008a
	Global	19	1	I			JRC/EUCAR/CONCAWE, 2008
JATROPHA	India	1.42	577	514		Husks and cakes as fertiliser or for fuel	Reinhardt et al., 2007
	Global	I	2.000	1.780	I	Detoxified meal or cakes as animal feed	Fulton et al., 2004
						Meal to produce biogas Shell as fuel Glycerol	
WASTED VEG- ETABLE	Global	I	I	1	[4.8–5.8]	Meal or cakes as animal feed	Rajagopal et al., 2007 in FAO, 2008a
OIL				П1 	E		
Biomass feedstock	Countries regions	Yields m <sup>3</sup> t FM ha <sup>-1</sup>	Biofuels m <sup>3</sup> ha <sup>-1</sup>	Fossil equivalent (on energy basis) $m^3 ha^{-1}$	Iotal energy output/Fossil energy input	Co-products	References
DIGESTIBLE MATERI-			Biogas	(% of CH <sub>4</sub> ) m <sup>3</sup> methane			
ALS				(LHV: 30 MJ m <sup>-3</sup>			
MAIZE	Germany	170–202	7.800-8.300	(52) 4.000– 4.300			(FNR, 2008)
	France	200	6.660-8.500	(52) 3.500- 4.450	I		www.pleinchamp.com (03.03.08)
GRASS	France	211	2.000-2.600	(54) 1.100– 1 400			Poiret, 1996
SILAGE	Germany	172	2.500–3.200	1.400 (54) 1.400– 1.700			Moras, 2007
SLURRY							
Bovine	Germany		$25 \text{ m}^3.\text{t}^{-1}$	(60) 15 $m^3 t^{-1}$			
	Global		$26 \text{ m}^3 \text{.t}^{-1}$	(60) 15.6 m <sup>3</sup> .t <sup>-1</sup>			

Pig	Germany	$28 \text{ m}^3.\text{t}^{-1}$	(65) 18.2 $m^3 t^{-1}$	I		
	Global	$36 \text{ m}^3.\text{t}^{-1}$	(65) 23.4 $m^3 t^{-1}$			
Chicken	Global	$44 \text{ m}^3.\text{t}^{-1}$	(60) 26.4			
MANURE			m <sup>3</sup> .t <sup>-1</sup>		CO <sub>2</sub> (25–45% of biogas) Digested biomass as fertiliser	
Bovine	Germany	$45 \text{ m}^3.\text{t}^{-1}$	$(60) 27$ $m^3.t^{-1}$			Moras, 2007
	Global	$25 \text{ m}^3.\text{t}^{-1}$	(60) 15 m <sup>3</sup> t <sup>-1</sup>			FNR, 2008
Pig	Germany	$60 \text{ m}^3.\text{t}^{-1}$	(60) 36 m <sup>3</sup> .t <sup>-1</sup>	I		
Chicken	Germany	$80 \text{ m}^3.\text{t}^{-1}$	(60) 48 m <sup>3</sup> .t <sup>-1</sup>			
BIOMASS WASTE						
Green waste	Germany	$100 \text{ m}^3.t^{-1}$	(61) 61 $m^{3} t^{-1}$			
	Global	$60 \text{ m}^3.\text{t}^{-1}$	(61) 36.6 m <sup>3</sup> .t <sup>-1</sup>			
		$151 \text{ m}^3.\text{t}^{-1}$	(61) 92 m <sup>3</sup> .t <sup>-1</sup>	I		
Maize straw	Global	$402 \times 10^{3} L.t^{-1}$	I			
WASTED OILS	Global	$800 \times 10^{3} L.t^{-1}$	I	I		
MISCANTHUS (>3 years)	Germany	15 6.081 L ha <sup>-1</sup>	4.044 L ha <sup>-1</sup> gasoline			Nguyen et al., 2007
CEREAL	2nd generation	6 2.390 L ha <sup>-1</sup>	4.590 L ha <sup>-1</sup> gasoline	20		FNR, 2007
STRAW OR WOOD–BTL	biofuels stand	- 4.030 L ha <sup>-1</sup>	$3.900 \text{ L ha}^{-1}$ diesel	I		
Notes: energy rat FM: fresh matter;	CCDS, corn conder	et and range given red distiller's solubles;	n in Worldwatch Ir DDGS, dried distil	stitute (2007), except thos ler's grains with solubles.	e in italics.	



Fig. 3 Worldwide ethanol and biodiesel productions in 2007 in Million litres; region shares (%). Drawn from data of F.O. Licht<sup>13</sup> in FAO 2008a

countries in Asia and Oceania, and more than tripled in Brazil. Figure 3 show the production shares of ethanol and biodiesel worldwide in 2007.

Despite the growth in biofuel consumption and a comparatively slower growth in oil consumption, biofuels still do not represent a significant share in worldwide liquid fuel supply; about 0.9% by volume, 0.6% by transport distance travelled (Worldwatch Institute, 2007). Within Europe, biofuels are essentially domestically produced and consumed, except in Sweden, where since 2004 all petrol has been blended with 5% ethanol, mainly originating from Brazil and wine production sites in Southern Europe; only 20% is produced nationally (Van der Drift and Boerrigter, 2006). International trade in biodiesel is minimal as yet:

only 10% of biofuels produced in the world are sold internationally, with Brazil accounting for roughly half of those sales (IEA, 2006). However, trade is very likely to increase, notably due to the fact that only some regions of the world may be able to produce large biomass feedstock (see part 6).

# 3.3 Towards 2nd and 3rd Generations of Biofuels

Second generation biofuels are produced via biochemical (hydrolysis and fermentation) and thermochemical (pyrolysis or gasification) treatments. The biochemical or so-called "wet process" is very similar to the 1st generation ethanol except for the feedstock, which is not specific. Indeed, 2nd generation biofuels are all produced from lignocellulose, i.e. all

<sup>&</sup>lt;sup>13</sup> Christoph Berg is Managing Director at the commodity analysts F.O. Licht. F.O. Licht monitors the global soft commodity markets.

kinds of vegetal biomass, as lignocellulose forms the basic structure of vegetal cell walls. Cell walls make up a substantial portion of the dried biomass: about 60–80% and 30–60% in the stems of woody and herbaceous plants, respectively, and about 15–30% in their leaves (Möller et al., 2007).

Lignocellulose consists of intricate assemblages of cellulose, hemicellulose and lignin, whose proportions and molecular organisation vary depending on the type of biomass. A typical range is 40 to 55% cellulose, 20 to 40% hemicellulose, and 10 to 25% lignin (Worldwatch Institute, 2007). The other minor components of cell walls are proteoglycans and pectins that glue together all the lignocellulosic compounds. The conformation of glucose residues in the crystalline cellulose core of cell-wall microfibrils forces the hydroxyl groups into radial orientation and the aliphatic hydrogen atoms into axial positions. It leads to the creation of strong interchain hydrogen bonds between adjacent chains in a cellulose sheet, which make cellulose resistant to enzymatic hydrolysis, and weaker hydrophobic interactions between cellulose sheets that contribute to the formation of a water layer near the hydrated cellulose surface, protecting cellulose from acid hydrolysis. Furthermore, the microfibrils are embedded in the matrix of hemicelluloses and lignin, the latter also contributing to making cells walls hydrophobic and more resistant against enzymatic attack (Möller et al., 2006). Other molecules such as, for instance, waxes or inhibitors of fermentation, which naturally exist in the cell walls or are generated during conversion processes, also contribute to biomass recalcitrance (Himmel et al., 2007). This recalcitrance is the primary barrier to producing ethanol from lignocellulosic feedstock, commonly referred to as cellulosic ethanol. Indeed, expensive pre-treatments are necessary to break down this resistance, and reaching a cost-effective cell wall saccharification, i.e. the degradation of cell walls into monosaccharides, is the key that could really permit cellulosic ethanol to enter the market.

Pre-treatments include physical methods such as milling and grinding, high-pressure steaming and steam explosion, and biological (lignin- or/and cellulose-degrading organisms) or chemical methods (alkali or acid treatments, solvents) to solubilise parts of the hemicelluloses and the lignin. So far, methods such as ammonia fibre explosion (AFEX), wet oxidation and liquid hot water (LHW) treatment seem to be more successful for agricultural residues, whereas steam pre-treatment has resulted in high sugar yields for both forestry and agricultural residues (Hahn-Hägerdal et al., 2006). Monosaccharides from cellulose (glucoses) and hemicelluloses (pentose sugars) are then released through acid- or enzyme-catalysed hydrolysis, and finally fermented. Concentrated or dilute acid hydrolysis methods are more mature but very energy-intensive and present the disadvantage of potentially also degrading the monosaccharides. Enzymatic degradation, on the contrary, is more specific and perceived by many experts as a key to cost-effective saccharification, but none of these methods is currently cost-effective (Möller et al., 2006). As an example, hydrolysis of pre-treated lignocellulosic biomass requires one hundred-fold more enzymes than hydrolysis of starch (Tolan, 2006 in Möller et al., 2006).

Researchers have therefore been focusing, on the one hand, on improving the yields of pre-treatment and lowering their costs, and on the other hand, on developing integrated processes that make it possible to protect, separate and use the other compounds, such as the C5-sugars and the lignin, or the co-products such as furfural and fermentation gases. Traditional fermentation processes relied on yeasts and microbes that only convert C6-sugars (mainly glucose) into ethanol (Fulton et al., 2004). Researchers have already succeeded in producing several new yeast strains and bacteria, such as engineered *E. coli, K. oxytoca* and *Z. mobilis* (Balat et al., 2008), that exhibit varying degrees of ability to convert the full spectrum of available sugars into ethanol.

The introduction of simultaneous saccharification and fermentation (SSF) permitted a gain in efficiency of a 13% higher overall ethanol yield than separate hydrolysis and fermentation (SHF) (72.4% versus 59.1% of the theoretical maximum yield) (Ohgren et al., 2007). This gain is due to the fact that hydrolysed sugars are immediately fermented in the case of SSF, whereas their accumulation leads to enzyme inhibition in the case of SHF (Fulton et al., 2004). More recently, the simultaneous saccharification and co-fermentation of hexoses and pentoses (SSCF) has proved to be further advantageous as the hexose sugars continuously released by enzymatic hydrolysis increase the rate of glycolysis, so that pentose sugars are fermented faster and with higher yield (Hahn-Hägerdal et al., 2006). This makes it possible to lower the cost as both operations can be done in the same tank, added to the

fact that enzyme manufacturers have recently reduced costs substantially thanks to biotechnology (Solomon et al., 2007; Balat et al., 2008). Nevertheless, further advances in discovering new hydrolases, new fermentation enzymes and organisms with process-tolerant traits such as tolerance to alcohol, pH and inhibitors, and advances in product recovery technology are required to reach commercial viability (US DOE, 2006a). Producing enzymes with combined tolerant traits is a real challenge, considering, for instance, that the majority of organisms cannot tolerate ethanol concentrations above 10-15% (w/v) (Balat et al., 2008). Moreover, optimal temperature and pH conditions vary depending on the enzymes and microorganisms involved in the different process stages, which can hamper the efficiency of the batch SSF or SSCF (Cardona and Sanchez, 2007; Öhgren et al., 2007). "Consolidated biomass processing" (CBP), the logical end point in the evolution of biomass conversion technology, would require a unique microbial community to produce all the enzymes for the saccharification and fermentation within a unique reactor vessel, but it has not been achieved yet (Fulton et al., 2004). For robust and complete conversion of polysacchaarides locked in biomass, the ultimate ethanologens will have to produce at least a dozen enzymes of different catalytic activities. Engineering such a yeast strain requires (1) screening thousands of combinations of biomassdegrading enzymes to identify the appropriate set of enzymes, then (2) managing to ensure that this strain is capable of simultaneously expressing the genes for all necessary enzymes (Hector et al., 2008).

Current biomass-conversion technologies were developed empirically and are based on limited understanding of the biological and chemical properties of biomass (Himmel et al., 2007). Therefore, all research efforts also rely in parallel on fundamental research to understand and characterise the cell walls of a very wide range of biomass feedstock better. This feedstock encompasses perennial grasses, short rotation coppices, cereal straws, and other biodegradable residues or waste. According to Möller et al. (2007), poplar, willow, miscanthus (see picture below) and wheat straw are the main relevant feedstock in Europe. In the US, attention is especially paid to maize stover, wood waste and switchgrass, whereas sugar cane producers are obviously more interested in converting the sugar cane bagasse. Research worldwide includes breeding programmes to develop new varieties with interesting

phenotypes in terms of growth and resistance, but also in more specific biorefinery-related terms, i.e. regarding the cell wall composition. Research also includes genetic engineering. As an example, "Spartan maize" has been genetically modified within a research programme at the Michigan State University to express cellulase and hemicellulase in the plant's leaves and stover (Sticklen, 2007). Transgenic alfalfa has also demonstrated lower amounts of lignin, leading to drastic reduction of pre-treatment costs (Chapple et al., 2007). However, reduction in lignin content also leads to reduced biomass by up to 40%, which emphasises the need to determine whether cell wall manipulation may compromise the plant's structural integrity or susceptibility to pests and pathogens (Chapple et al., 2007).



Miscanthus at Estrees-Mons, November 2008 bessou©INRA

Figure 4 shows the expected time frame for research advances and economically viable implementation of cellulosic ethanol over the next 5 to 15 years in the US. Considering the potential to sustainably harvest more than 1.3 billion metric tons of biomass from U.S. forest and agricultural lands by the mid-21st Century



Fig. 4 Phased development of bioenergy systems over the next 5 to 15 years (US DOE, 2006a). SSCF: simultaneous saccharification and co-fermentation; CBP: consolidated biomass processing

(Perlack et al., 2005), these projections illustrate the needed co-increase of technological performances and systems integration, and detail the research fields. Within 10 years, dedicated energy crops with composition and structure better suited for breakdown into sugars for fermentation, high yield and robustness will be essential in contributing to achieving energy security.

While large deployment may not occur before ten years (US DOE, 2006a; BIOFRAC, 2006; FNR, 2006), or even fifteen years (Möller et al., 2007), several pilot and demonstration plants have already been built worldwide. Some twenty of such plants have been implemented since 1985 in the US, Canada, Brazil, Europe and Japan and about a dozen cellulosic ethanol commercial plants were being developed in 2007–2008, essentially in the US (Solomon et al., 2007) or under discussion in Canada and China. Steam pre-treatment with the addition of a catalyst for hydrolysis and improved enzymatic digestibility is the closest technology to commercialisation and has been widely tested in equipment (Hahn-Hägerdal et al., 2006). pilot-scale Considering the state-of-the-art technology in 2006, an estimated capital investment for a 220-million-litre cellulosic ethanol plant would approximate US \$ 300 million, with the largest capital components for feedstock pre-treatment (17%) and simultaneous saccharification and fermentation (15%), and energy utilities (36%). The production cost could then approximate US \$0.57 per litre with 40% related to the annualised capital charge and 46% to the feedstock and other raw materials (Solomon et al., 2007). In another estimate, production and transport of feedstock would represent about 21% and 26% of the total annual plant costs, respectively (Kaylen et al., 2000). In recent simulations, production costs, mostly based on the laboratory scale, range from 0.28 to 1.0 US \$ per litre of cellulosic ethanol (Hahn-Hägerdal et al., 2006). In 2006, the production cost of dry mill ethanol from maize was US \$ 0.44 per litre (Balat et al., 2008).

Scientists mostly argue that the technology is not mature yet for commercial production (Solomon et al., 2007; Cardona and Sanchez, 2007), whereas some industrials may be ready to take the chance. Still, all agree that tremendous increase in production volumes is the determinant techno-economic factor to reach commercial viability. About 86% of operating costs appeared to be proportional to the size of the plant (Kaylen et al., 2000). A drastic increase in production volumes and an "on-site" enzyme production, provided with governmental funds in the first development phase, appear to Murray Burke<sup>14</sup>, Vice President and General Manager of the SunOpta BioProcess Group, the essential challenges to reach commercial production. As costs are highly linked to feedstock, whose price is volatile, diversification of feedstock, maximisation of ethanol yields and optimisation of the use or commercialisation of co-products must also be achieved. Current pilot plants can produce a few million litres a year, possibly integrated with an ethanol from-grain plant, which can be a near-term solution (Hahn-Hägerdal et al., 2006), but a ten-fold capacity increase appears to be necessary for a complete switch of these plants to cellulosic ethanol plant.

Whereas lignin cannot easily be converted through biochemical processes, it can be burnt. Therefore, thermochemical processes are especially more effective in the case of plants with a high content of lignin, up to 30-35% of the biomass in some plants (Fulton et al., 2004; Möller et al., 2006). The main criteria for biochemical conversion of lignocellulosic feedstock are the quantity of sugars and the structure of the lignocellulose; in the case of thermochemical conversion, the main criteria are rather the biomass' bulk density, moisture and ash contents, and the calorific value (Möller et al., 2007). In a rough overview, agricultural residues and grasses with intrinsically higher sugar content and lower lignin content are generally more suitable for enzymatic conversion, whereas dense woody biomass with higher amounts of lignin and lower amounts of ash is comparatively more oriented toward thermochemical conversion. Ash can indeed lead to the slagging or fouling of heat-transfer surfaces during gasification. However, improvement of current technologies will notably permit one to reach efficient

conversion ratios for a mix of the cheapest and most available feedstock within the supply area of an implemented technology (Worldwatch Institute, 2007).

The thermochemical pathway is referred to as Biomass to Liquid (BtL) as an analogy with the conventional fossil Gas to Liquid pathway (GtL). Nowadays, 8% of the syngas produced worldwide is converted into transportation fuels through GtL processes; the overall production of syngas corresponding to almost 2% of the total worldwide primary energy consumption. Thereby, thermochemical technologies are well developed but have to be adapted to biomass feedstock in qualitative terms as well as in terms of plants' scale, considering that biomass availability might appear to be a limiting factor (Van der Drift and Boerrigter, 2006). The core process is gasification, that involves using heat to break down the feedstock molecules and produce a synthetic gas or syngas (also called "bio-syngas" when biomass is the feedstock), and whose compound-mix of hydrogen, carbon monoxide and dioxide, water vapour, methane and nitrogen varies depending on the process (Fulton et al., 2004).

There are two major types of gasifier that were selected because of their high efficiency in producing H<sub>2</sub> and CO, although they still produce a different ratio of gases at different temperatures and with a differing level of cleanliness (ITDG, 2000; Van der Drift and Boerrigter, 2006). The fluidised bed gasifier, typically operated at 900°C, has already been developed and demonstrated to produce heat and power from biomass. It requires, though, a catalytic reformer downstream to treat the produced gas so that it can fulfil the requirements to be converted into biofuels. On the contrary, the entrained flow gasifier, typically operated at 1300°C, makes it possible to produce syngas without a catalytic reformer but needs energyintensive pre-treatment of the biomass, such as torrefaction or pyrolysis, in order to reach a sufficient conversion rate. Torrefaction at 250-300°C or flash/slow pyrolysis at 500°C both turn solid biomass into a biocoal or a bio-oil/char, respectively, that can be easily transported and fed to the gasifier. In all processes, syngas has to be further conditioned via gas cleaning and the  $H_2/CO$  ratio adjusted to be fed to a synthesis reactor and converted into final biofuels, such as Fischer-Tropsch diesel and naphta (basic gasoline), dimethyl ether (DME), methanol, mixed alcohols or hydrogen. Hydrothermal upgrading (HTU) is another

<sup>&</sup>lt;sup>14</sup> Speech at the Platts Cellulosic Ethanol Conference in Chicago on October 31, 2006.

process that makes it possible to transform biomass into a "biocrude" liquid by dissolving the cellulosic materials in water under high pressure but relatively low temperature. Bio-oils produced via pyrolysis or HTU can be subsequently upgraded into diverse hydrocarbon liquids and fuels (Fulton et al., 2004).

The diverse options are not incompatible; fluidised bed gasification can even occur as pre-treatment to feed an entrained flow gasifier. Choices are guided by the type of available biomass and the desired biofuels. Research projects are currently focusing on improving the pre-treatments, adapting the scale, and developing integration options such as polygeneration and mature biorefinery concepts to make the processes economically viable. Polygeneration refers here specifically to the up-value of all produced gases, and in particular of methane used as biogas to provide heat and power. Mature biorefinery (Fig. 5) is the combination of biochemical and thermo-chemical treatments that will permit one to produce as much biofuel (54% ethanol from sugars, 10% FT-diesel and 6% FT-gasoline from lignin and other residues) and heat/power (5% electricity also from lignin and other residues) as possible from lignocellulosic biomass (100%) with a minimum of energy input (21% captured for process energy or lost and <5% agricultural inputs: e.g. farming costs, feedstock transport) (US DOE, 2006a).

The frontier between 2nd and 3rd generation biofuels is conceptual and is not due to differences in biomass feedstock or radically new conversion processes. Still further technological breakthroughs will be needed to permit the economic viability of completely integrated biorefinery complexes, as well as technological revolutions in the transportation sector to introduce hydrogen as a competitive fuel for automotives. Hydrogen (H<sub>2</sub>) is a fuel whose combustion produces only water. Although water vapour is the most significant greenhouse gas, its equilibrium in the atmosphere seems to be ensured by the natural water cycle. Hydrogen has been used by the aerospace industry since the 1960s and is nowadays



**Fig. 5** Mature biomass refining energy flows: example scenario in US DOE, 2006a. "Envisioning Mature Biomass Refineries," presented at First International Biorefinery Symposium, Wash-

ington, DC (July 20, 2005). Dotted arrows from above indicate energy inputs needed to run machinery

especially used in the petrochemical industry to make ammonia fertilisers, to upgrade lower quality fractions in the refining of fossil fuels, and also to produce glass, lubricants, refined metals and processed foods (Zeman, 2007). According to Shell, the world market for distributed and centralised hydrogen is estimated at approximately 45 million tons per year. However, hydrogen is not to be found in nature under this diatom form and has to be produced from hydro-carbonates or water, requiring considerable energy inputs.

Hydrogen is designated as a 3rd generation biofuel, when it is produced from biomass via the thermochemical processes described above. However, this term would not be appropriate when talking about hydrogen coming from the conversion of fossil fuels, even if the processes ensured the storage of all emitted greenhouse gas during the conversion. 95% of today's hydrogen is produced from fossil combustible, most commonly methane (Demirbas, 2007), via steam reforming that releases CO2 into the atmosphere. Production via water electrolysis is three to four times more expensive and has a low energy yield (CEA, 2004). Hydrogen from renewables for fuel celldriven vehicles might be a long-term solution, but its introduction needs breakthroughs in technology and cost and would require intermediate steps, to make a gradual growth of both fuel availability and number of vehicles possible. An effective intermediate step will be the use of hydrogen as a component in fuel production processes from biomass. This is applicable for today's fuel routes via synthesis gas, but will also be a serious option for future biorefineries (BIOFRAC, 2006). The development of a profitable hydrogen chain will take longer, especially considering the gas' inherent limit in terms of compression and storage on board. Although hydrogen contains three times as much energy as gasoline per unit weight, 4.6 litres of hydrogen compressed at 700 bars are needed to substitute 1 litre of gasoline (CEA, 2004). Moreover, as it is a flammable very small molecule, it requires specific hydrogenproof material to be stored and transported. Currently hydrogen transportation is 50% more expensive than natural gas transportation, notably because one volume unit of hydrogen contains three times less energy than the same volume unit of natural gas (CEA, 2004).

Considering the risks and following costs implied in the development of new biofuel chains, industries' investments are significantly subordinated to the commercial perspectives that global policies underpin. These policies tend to respond to global issues and inevitably affect trade, as economic incentives often appear as efficient levers to reach targets.

#### 4 Political and Economic Frameworks

## 4.1 Climate Change and Greenhouse Gas Emission Trends

Lately biofuels have been fostered worldwide in a double context of energy insecurity and climate change. Except for a few exceptional cases, such as the Brazilian Pró-álcool Programme launched in 1975, it was not until the awareness of the risks associated with the depletion of fossil resources was drastically raised that biofuels and other renewables were widely given attention as real potential energy sources. Since the late 1980s, the more explicit the conclusions of the Intergovernmental Panel on Climate Change on the reality of climate change and the impact of anthropogenic greenhouse gas emissions<sup>15</sup> have become, the more concrete the international policies and instruments to promote renewables have appeared. Needs for action and cooperation have been expressed within the framework of international agreements; such as the Framework Convention on Climate Change (UNFCCC) in 1992 and the Kyoto Protocol in 1997. Although they might not have federated enough stakeholders, notably the Kyoto Protocol which only entered into force in 2005 without some of the main CO<sub>2</sub> contributors, they gave way to the establishment of effective frameworks and national action plans.

The global average surface temperature on the Earth increased about  $0.7^{\circ}$ C between the late 1800s and 2000, with a rate of about  $0.2^{\circ}$ C per decade (IPCC, 2007) in the past three decades. However, taking into account the effects of orbital variations on climate, absent human influence, the natural trend would be toward a cooler climate, as peak warmth of the current interglacial period (Holocene) occurred 8–10 thousand years ago. Examination of prior interglacial periods reveals a strong correlation between the CO<sub>2</sub> and CH<sub>4</sub> concentrations in the

<sup>&</sup>lt;sup>15</sup> By default examining greenhouse gas emissions includes "six" gases: CO<sub>2</sub>, CH<sub>4</sub>, N<sub>2</sub>O, SF<sub>6</sub>, PFCs and FCs.

atmosphere and temperature records. Nevertheless, in the past the temperature changes usually preceded the changes in gas concentrations. Today, anthropogenic greenhouse gas emissions are overwhelming and have reversed the order so that greenhouse gases are driving temperature increases. The climate system has not come to equilibrium with today's climate forcing and more warming is "in the pipeline". "Humans now control global climate, for better or worse" (Hansen, 2006). In other words, the IPCC stated in its last report: "Most of the observed increase in global average temperatures since the mid-20th century is very likely (probability >90%) due to the observed increase in anthropogenic greenhouse gas concentrations. Discernible human influences now extend to other aspects of climate, including ocean warming, continental-average temperatures, temperature extremes and wind patterns" (IPCC, 2007). Pre-industrial global atmospheric concentrations of CO2, N2O and CH4 have increased markedly as a result of human activities since 1750<sup>16</sup> and now far exceed pre-industrial values determined from ice cores spanning many thousands of years.

Global increases in CO<sub>2</sub> concentration are due primarily to fossil fuel use and land-use change, while those of CH<sub>4</sub> and N<sub>2</sub>O are primarily due to agriculture. If CO<sub>2</sub> emissions continue to increase by 1.5 to 2% per year, doubled CO<sub>2</sub> will be reached in approximately the year 2050. Encompassing the whole range of the six IPCC emission scenarios<sup>17</sup> from the lowest to the highest emissions, global warming could reach 1.8 to 4°C by 2100 (IPCC, 2007). A global warming of 2 to 3°C over the pre-industrial temperature would already "make the Earth a different planet" (Hansen, 2006). As a very critical issue, sea level rise illustrates how climate change can lead to exponential and irreversible impacts due to accumulation phenomena and positive feedbacks. IPCC scenarios give estimates of a sea level rise between 38 cm and 59 cm by the end of the 21st century relative to 1980–1999, due mostly to thermal expansion and excluding future rapid dynamical changes in ice flow. There is still no consensus on the long-term future of the ice sheet or its contribution to sea level rise. It is not possible to say how long it would take for sea level to change as feedbacks can

<sup>16</sup> Pre-industrial concentrations/in 2005: CO<sub>2</sub> (280/379 ppm); N<sub>2</sub>O (270/319 ppb); CH<sub>4</sub> (715/1774 ppb), IPCC, 2007. lead to highly non-linear responses, nevertheless "it is almost inconceivable that under the business-as-usual scenario climate change would not yield a sea level change of the order of meters on the century timescale" (Hansen, 2007). Given the populations in 2000, a sea level rise of 6 m would displace 35 million inhabitants throughout the world and trouble is brewing for many species.

The distance that climate zones have moved so far is small, but the rate of movement of isotherms is now pole-ward at 50 km per decade and will double this century if we follow the business-as-usual scenario, surely causing the extinction of lots of species (Hansen, 2006). The IPCC stresses that: "Continued greenhouse gas emissions at or above current rates would cause further warming and imply many changes in the global climate system during the 21st century that would very likely be larger than those observed during the 20th century. [...] Sea ice is projected to shrink in both the Arctic and Antarctic under all scenarios<sup>18</sup>. It is very likely that hot extremes, heat waves and heavy precipitation events will continue to become more frequent. Even if the concentrations of all greenhouse gases and aerosols had been kept constant at year 2000 levels, a further warming of about 0.1°C per decade would be expected" (IPCC, 2007). An alternative scenario aims at limiting the  $CO_2$  peak at 475 ppm in 2100 before it should slowly decline thereafter and also requires a reduction of non-CO2 forcing gases in order to hold warming to less than 1°C. The 500 ppm scenario could make it possible to hold warming to less than 2°C. From today's perspective, the 2°C target is only achievable if global emissions are reduced below 10 GtCO<sub>2eq</sub> yr<sup>-1</sup> in the longer term, meaning more than halving the 1990 level. In 2004, 51 GtCO<sub>2eq</sub> were added to the atmosphere, and the rise in 1990 emissions alone also produced an additional annual steady flow of 39 GtCO<sub>2</sub> due to climate time-lagged response to greenhouse gas emissions. If this development continues, it will be impossible to stay within the aforementioned limit for temperature increase (Fischedick et al., 2007); another decade of business-as-usual would eliminate the Alternative Scenario (Hansen, 2006).

<sup>&</sup>lt;sup>17</sup> SRES: special report on emission scenarios IPCC.

<sup>&</sup>lt;sup>18</sup> Sea ice melting does not directly cause sea level rise like ice on continents; however, it can lead to the extinction of species that rely on these relatively scarce habitats. It also contributes to ocean thermal expansion.
Together, the 25 countries with the largest greenhouse gas emissions account for approximately 83% of global emissions. The largest emitter is the United States, with 21% of global emissions, followed by China with 15%. It follows that most of the remaining countries contribute little to the build-up of greenhouse gases in the atmosphere; 140 countries contribute only 10% of annual emissions (Baumert et al., 2005). The largest percentage increase since 1976 occurred in 2004, when more than 28 GtCO<sub>2</sub> were added to the atmosphere from fossil fuel combustion alone. Emission growth rates are highest among developing countries, where collectively CO<sub>2</sub> emissions increased by 47% over the 1990 to 2002 period. Among the major developing country emitters, growth was fastest in Indonesia (97%), South Korea (97%) and Iran (93%). During the same period, emissions also increased mainly in Canada (+20%) and Australia (+22%), whereas emissions in most developed countries did not change. During the 2003–2004 period, the CO<sub>2</sub> growth of 50% in China accounts for more than half of the worldwide CO<sub>2</sub> increase.

The Kaya Identity<sup>19</sup> model (Fig. 6) gives some clues to understanding the energy-related  $CO_2$  emissions by using four factors. Emission intensity or "carbon intensity" is a function of "energy intensity" and "fuel mix". Energy intensity reflects both the level of energy efficiency and the overall economic structure of a country. An economy based on heavy industrial production, for instance, is more likely to have higher carbon intensity than one where the service sector is dominant. However, energy-intensity levels are not



Fig. 6 Kaya Identity model

well correlated with economic development. In several countries, it can be seen that declines in intensity were accompanied by significant increases in GDP, leading to increases in absolute  $CO_2$  levels. The most notable case is China, where the effect of significant intensity declines, although China still heavily relies on the coal industry, was more than offset by substantial GDP growth. Likewise, the US decline in carbon intensity

(17%) was offset by increases in population and GDP,

giving a significant greenhouse gas emission growth

in the US of 13% over the 1990 to 2002 period.

Whereas in agrarian economies with little heavy industry or energy production, land-use change, especially tropical deforestation, represents a larger share of CO<sub>2</sub> emissions, in a majority of countries, economic growth has, finally, the strongest influence on emission levels. This is the case in countries as diverse as the US, India, Australia and Iran (Baumert et al., 2005). Given the diversity of large emitting countries, it is simply not possible to adequately address the climate change problem without engaging both developed and developing countries, while adapting mitigation instruments to the specificity of influencing factors in the diverse countries. Fixed emission "caps" in particular may be impracticable in developing countries where economic growth is robust. Furthermore, differentiated per capita greenhouse gas emission targets rather than absolute emissions would reduce the effects of population growth on the commitments of Parties. Total emissions of CO<sub>2</sub> in 2004 show, for instance, China (3.7 tCO<sub>2</sub>/capita) being far below the US (19.6 tCO<sub>2</sub>/capita) and EU-27 (8.7 tCO<sub>2</sub>/capita) (EEA, 2008). Also, projections of carbon intensity tend to exhibit less uncertainty than absolute emission forecasts (Baumert et al., 2005).

Between 1990 and 2004, European global greenhouse gas emissions had decreased from most sectors, particularly energy supply, agriculture and waste management; except from transport, which increased by nearly 26%. Had transport sector<sup>20</sup> emissions followed the same reduction trend as in society as a whole, total European Union-27 greenhouse gas emissions during

<sup>&</sup>lt;sup>19</sup> Model developed by the Japanese energy economist Yoichi Kaya in Environment, energy, and economy: strategies for sustainability, co-authored with Keiichi Yokobori as the output of the 1993 Tokyo Conference on Global Environment.

<sup>&</sup>lt;sup>20</sup> The transport sector presented here consists of road transportation, domestic civil aviation, railways, national navigation and other transportation. It excludes emissions from international aviation and maritime transport (which are not covered by the Kyoto Protocol or current EU policies and measures). Road transport is by far the biggest transport emission source.

the period 1990-2005 would have fallen by 14% instead of 7.9% (EEA, 2008). Only Germany, France and Portugal have managed to stabilise transport emissions in recent years. Nevertheless, EU-15 greenhouse gas emissions from transport are still expected to increase a further 35% above 1990 levels by 2010 if only existing policies and measures are used (EEA, 2006a). The transport sector represents the most significant climate policy challenge at two levels. First, transport contributes the lion's share of the emission increase of the European Union and in spite of the voluntary agreement to reduce the carbon content of travel for new vehicles, there does not appear to be in the near future a technological solution of a magnitude that could offset the effect of increased traffic and increased onboard equipment on  $CO_2$  emissions. Second, further efforts to mitigate emissions in other sectors will be difficult to accept if governments do not undertake meaningful efforts in the transport sector (Barbier et al., 2004).

In the EU-25, despite an annual 1 t decrease in the average greenhouse gas emissions per capita of  $CO_{2e}$  between 1990 and 2004 and some successful decoupling of greenhouse gas emissions and economic growth, total EU-23 greenhouse gas emissions rose in 2004 by 0.3% compared with 2003 and were 5% below the 1990 level, the highest level since 1997. With existing policies and measures, and without additional ones, EU-23 greenhouse gas emissions are projected to keep on increasing and to be 2.1% below the 1990 level by 2010, meaning that the EU-15 Kyoto commitment of 8% emission reduction from this base-year level by 2008–2012 would not be reached, although the eight new member states had in 2004 emissions of only 76.8% of those in 1990 (EEA, 2006a).

At the UNFCCC meeting in Bali, December 2007, representatives of 180 countries agreed on a 'Bali roadmap' with the aim to achieve by the end of 2009 a global post-2012 climate change agreement to limit emissions, and address other issues such as adaptation to climate change, after the end of the Kyoto Protocol commitment period (2008–2012), whose targets will not be achieved. It should include both developed and developing countries, but with the largest emission reduction effort expected by the developed countries (indicatively in the range of 25 to 40% emission reductions by 2020 from 1990 levels). The European Council agreed in March 2007 on an integrated energy

and climate change strategy. It endorses an EU objective of a 30% reduction in greenhouse gas emissions by 2020 compared with 1990 levels, provided that an international agreement can be reached with other industrialised countries. Without such an agreement, the EU would still pledge to a firm independent commitment to achieve at least a 20% reduction (EU, 2007). The EU Commission proposed to split the overall emissions reduction target into two: one for the sectors covered by the European Emissions Trading Scheme (ETS) and one for the non-trading sectors in which transport is included (EEA, 2008).

# 4.2 Biofuel-Related Policies

#### 4.2.1 European Policies

In Europe, several directives have been released; notably in 1997, the Energy White Paper and Action Plan, and the Green papers on Energy Supply (2000) and Energy Efficiency (2005). These policy instruments notably set indicative objectives in terms of consumption of renewables, e.g. on the share of green electricity (EU, 2001), the promotion of biofuels or emissions trading (EU, 2003), and the use of waste and disposal (2005). In parallel, other directives were released, which notably deal with energy efficiency improvement, economic incentives and eco-labels. However, none of the given targets were binding ones and the results have so far not been convincing. The White Paper (CEC, 1997) on the share of RES in total energy had proposed a common framework for action aiming to achieve the indicative objective of 12% for the contribution of renewables to the EU gross inland energy consumption by 2010, i.e. to double the share of renewables compared with 1997, including a tripling of biomass use. In 2003, the total amount of renewables used averaged only 6% of the EU gross inland energy consumption (EU DG-TREN, 2006a), about two-thirds of this contribution coming from biomass, i.e. 4% of EU total energy needs (CEC, 2005). Even if renewables consumption can widely differ between Member States, the challenge to reach the global objective remains entire; only 7% of the necessary growth of bioenergy production has been achieved globally (Fagernäs et al., 2006). In 2005, the indicative target<sup>21</sup> of 2% market share for biofuels stated in the EU Council Directive on "Biofuels" (EU, 2003) was not reached; biofuels apparently merely attained 1.4% of market share within the EU-25 (EU DG-TREN, 2006b). This share was better than in 2003 (0.6%) but if this trend continues, the 2010 target of 5.75% share will not be achieved: the forecast indicates a 4.2% share by 2010 (CEC, 2008).

Considering the need for a drastic reduction of greenhouse gases from transport and the still very low incorporation of biofuels, the European Union has decided to put into force a new directive that fixes mandatory targets, as was agreed during the European Union Summit in March 2007: 20% of the global energy consumption of the European Union has to be renewably sourced, including a minimum binding target of 10% within the transport sector (only consumption of gasoline and diesel are considered) for each Member State by 2020 (CEC, 2008). This directive proposal was published in January 2008 and should be followed up by concrete Member States' action plans by the end of March 2010. It will replace the former 2001/77/CE and 2003/30/EC directives after the 31st of December 2011. As a further incentive for investors, the Directive indicates that the share contribution by 2nd generation biofuels to the 10% binding target would count for double in comparison with the other biofuels (Article 18). For biofuels and other bioliquids, the directive sets up three conditional criteria of sustainability for their production, so that their consumption can be taken into account to fulfil the binding target and allow financial support (Article 15), although no sanction has been planned for non-compliance with these criteria or the non-fulfilment of both targets:

- a minimum saving of 35% of greenhouse gas emissions compared with the substitute fossil fuels,
- biomass feedstock must not be produced on soils within ecosystems considered to have a high value in terms of biodiversity: i.e. undisturbed forests, protected areas and specific permanent grasslands

that shall be geographically identified by the Commission,

 biomass feedstock must not be produced on soils with high organic carbon contents: specific humid areas, notably virgin peat soils, and forests wider than 1 ha with canopy covers superior to 30%.

Concerning agricultural feedstock, supplementary environmental criteria of the European regulation related to good agricultural practices (CE 1782/2003 Article 5 and Annex III point A) remain applicable. Member States shall require economic operators to show that the environmental sustainability criteria set out have been fulfilled. In particular, the Directive stipulates the method to calculate the greenhouse gas emissions throughout the production chain and states that emission reduction due to the co-products shall be handled either by system extension in the case of co-generated electricity from agriculture, or by energy allocation<sup>22</sup> in all other cases. It also gives minima of emission reductions for each biofuel chain (Annex VII) that shall serve as reference.

However, the method does not specify how to take into account the N2O emissions from the agricultural phase, and cannot at the Community level take into consideration the regional variability (see part 5). Furthermore, no independent certifying authority will be involved in the control of the respecting of the criteria, and biofuels are not included in the guarantee of origin system that will ensure the traceability of electricity, heating and cooling produced from renewable energy sources (Article 6-10). It is also mentioned that it would be technically and administratively unfeasible to apply EU environmental requirements for agriculture to biofuels and other bioliquids from third countries. In fact, since the proposal does not include any derogation for countries where the situations are constrasted with regard to continuous forest, peat soil or grassland resources as well as to certification access, establishing these biofuel sustainability criteria at a multilateral level could be perceived as a discriminatory measure according to the regulation of the World Trade Organization (Pons, 2008). Much remains to be done in order to establish multilateral agreements between the Community and third countries defining international standards of sustainability criteria, the greenhouse gas emission calculation method and certification control.

 $<sup>^{21}</sup>$  Minimum indicative targets from the European Council Directive 2003/30/EC of 8 May 2003: 2% in 2005 and 5.75% in 2010 share of biofuels of all petrol and diesel for transport purposes placed on the market calculated on the basis of the energy content. (about 3% and 8.6% for ethanol; 2.2% and 6.4% for biodiesel when calculated on a volume basis).

<sup>&</sup>lt;sup>22</sup> Based on the low calorific values.

#### 4.2.2 US Policies

For the last twenty years, the US government has also been putting into force several policies related to renewables in a view to reduce its dependency on imported oil. Starting with the Energy Tax Act of 1978, the US government has continuously maintained national tax incentives to encourage ethanol fuel production and use. Increases in ethanol excise tax exemption about every two years during the 80s-90s and loan guarantees to build up production facilities have notably fostered the growth of domestic maize-based ethanol (MacDonald et al., 2004). In 2000, the Biomass Research and Development Act directed the departments of Energy and Agriculture to integrate their biomass Research and Development and established the Biomass Research and Development Technical Advisory Committee (BTAC), which advises the Secretary of Energy and the Secretary of Agriculture on strategic planning for biomass Research and Development. In 2002, this Committee set up a challenging goal requiring biomass to supply 5% of the nation's power and 25% of its chemicals, and biofuels to meet 20% (10%) of transportation fuel consumption by 2030 (2020). These contributions would represent all together 30% of the current petroleum consumption (Perlack et al., 2005). The 2002 Farm Bill also established new programmes and grants that support increased use of biofuels and biobased products as well as advanced biorefinery development (US DOE, 2006a). The need to substitute MTBE<sup>23</sup>, which has been banned<sup>24</sup> in a growing numbers of US States due to its toxicity in high blends (formerly 15% in California, for instance), has also contributed to underpinning ethanol blends. However, biomass currently accounts for merely 4% of total energy consumption (BRDI, 2006), and biofuels, mostly maize ethanol, for around 2 to 3% of domestic transportation motor fuels (Kojima and Johnson, 2005; US DOE, 2006a).

With the growing energy consumption, US dependence on imported oil has reached severe levels. Between 1984 and 2005, crude oil imports increased 194%. In 2005, about 65% of crude oil and petroleum products were supplied by imports, representing 30% of the total US trade deficit. The overall demand for transportation fuels has increased 19% in the past ten years, with the vast majority of this growth reliant on imported petroleum (BRDI, 2006). During the last three years, the Government has hence especially insisted, with the Energy Policy Act of 2005, the American Competitiveness Initiative and the Advanced Energy Initiative (2006), on providing an aggressive strategy for tackling long-term energy challenges (US DOE, 2006b) and has shown ambitious

goals in terms of energy efficiency and diversification, i.e. the increase in domestic production of conventional fuels as well as the development of new nuclear power generation, hydrogen and renewables.

In terms of bioenergy, the Energy Policy Act of 2005 (EPAct), a US \$ 14-billion national energy plan, notably aims at fostering research programmes and partnerships between industries and academic institutions in order to develop advanced processes in bioproduct production. In this sense, it includes amendments to the Biomass Research and Development Act, focusing on four new technical areas for research and development activities: (1) develop crops and systems that improve feedstock production and processing, (2) convert recalcitrant cellulosic biomass into intermediates that can be used to produce biofuels and products, (3) develop technologies that yield a wide range of bioproducts that increase the feasibility of fuel production in a biorefinery, and (4) analyse biomass technologies for their impact on sustainability and environmental quality, security, and rural economic development.

The "Initiatives" strategies also particularly emphasise the role of technology development and innovations. The President's Advanced Energy Initiative provided a 22% increase in funding for clean-energy technology research at the Department of Energy in two vital areas: 1. "Changing the way we fuel our vehicles"; and 2. "Changing the way we power our homes and businesses". In 2007, a total budget of US \$150 million was allocated to the DOE to fund biomass research and help to reduce the costs of producing advanced biofuels and ready technologies for their commercialisation.

Finally, biofuel production objectives are also underpinned by renewable content requirements for

 $<sup>^{23}</sup>$  Methyl tertiary butyl ether: fossil oxigenate additive to gasoline.

<sup>&</sup>lt;sup>24</sup> Contamination of groundwater by MTBE due to leaking tanks is especially severe in the US. Despite the decision to phase it out, the quantities of MTBE used in the US have not decreased due to its technical advantages that actually help to produce a cleaner burning gasoline (http://www.acfa.org).

motor vehicle fuels. Called Renewable (or Alternative) Fuel Standard (RFS), the EPAct's provision requires gasoline sold in the US to be mixed with increasing amounts of renewable fuel on an annual average basis, up to at least 28 billion litres per year of biofuels by 2012 blended into the nation's fuel supply. In 2007, the US produced about 24.2 billion litres of ethanol and 1.7 billion litres of biodiesel, four times more ethanol than in 2000 and 80% more biodiesel than the previous year (US Government, 2008). In December 2007, President G.W. Bush signed the Energy Independence and Security Act, which notably responds to his "Twenty in Ten" challenge, a regulation to reduce gasoline consumption by 20% in ten years. This Act includes a new Renewable Fuel Standard, requiring fuel producers to supply at least 136 billion litres of renewable fuel in the year 2022, but also a Vehicle Fuel Economy Mandate, specifying a national mandatory fuel economy standard corresponding to a gain of 17 kilometres per litre by 2020. Several states in the US have adopted biofuel blend mandates; Louisiana, Montana, New Mexico, Oregon and Washington states, for instance, require ethanol (mostly 10%) in gasoline and/or biodiesel (2 to 5%) in highway diesel fuel with effective dates in the future. California is, moreover, developing a Low Carbon Fuel Standard for transportation fuels with a goal to reduce the carbon intensity by at least 10 % by 2020. US policies undoubtedly boosted the biofuel supply and were massively followed by the member states as illustrated by the Governors' Ethanol Coalition, that includes 32 member states out of 50, as well as international representatives from Brazil, Canada, Mexico, Sweden and Thailand (US DOE, 2006a).

Although biofuels were initially thought to contribute to lowering US energy dependence on imports, the US still imported about 1.7 billion litres of ethanol in 2007<sup>25</sup>. Even if all maize grain grown in the US were converted into ethanol, it would have satisfied just about 15% of the transportation needs (US DOE, 2006a). To reach the 30% vision of the BTAC, one billion dry tons of biomass feedstock would be needed annually, from the potential dedicated 1.3 billion dry tons from forestlands and agricultural lands, provided that large-scale bioenergy and biorefinery industries, including cellulosic ethanol plants, exist (Perlack et al., 2005). Therefore, the government especially focuses on the development of cellulosic ethanol (1 litre of non-grain-based ethanol is counted as 2.5 litres of grain-based ethanol to fulfil the RFS). This focus was confirmed when, on May 5, 2009, United States president Barack Obama signed a presidential directive on developing advanced biofuels, i.e. 2nd and 3rd gener-

developing advanced biofuels, i.e. 2nd and 3rd generation biofuels. Cellulosic ethanol could enable greater greenhouse gas savings, which appears to be a crucial means for the US to drastically lower its greenhouse gas emissions, while it is besides not particularly willing to commit itself within the international emission reduction targets plan.

#### 4.2.3 Chinese Policies

As Chinese economy and consumption levels boom, Chinese energy policy is likely to significantly affect the worldwide energy market and is fatally expected to play a growing and major role in greenhouse gas emissions. In 1975, China became a net oil importer. Today, it depends on coal for around 70% of its primary energy and the main role of coal within the energy structure will remain unchanged for a long time to come. Nuclear power and renewables account for about 7% of primary energy consumption, the rest comes from fuel oil used in the transportation sector, whose consumption is growing rapidly (SCIO, 2007). Considering that China's energy efficiency is about 10% lower than that of the developed countries, and its per unit energy consumption of energy-intensive products is about 40% higher than the advanced international level, priority is given to the up-grading and widening of the domestic energy-grid by implementing more efficient and cleaner technologies. The 11th Five-Year Plan for National Economic and Social Development of the People's Republic of China outlines that the per-unit GDP energy consumption by 2010 will have decreased by 20% compared with 2005, and the total amount of major pollutants discharged will have been reduced by 10 percent (SCIO, 2007).

In terms of renewable energies, national targets are to reach contributions of 10% and a further 15% of total energy consumption by 2010 and 2020 (SCIO, 2007). The Renewable Energy Law of the People's Republic of China<sup>26</sup> (09/11/2005, Article 16) as well as China's Energy Conditions and Policies

<sup>&</sup>lt;sup>25</sup> http://www.worldwatch.org (25/04/2008).

<sup>&</sup>lt;sup>26</sup> Online: http://www.ccchina.gov.cn/en/Public\_Right.asp? class=17.

(28/12/2007, pages 17 and 37) directly but briefly endeavour the production of biofuels. The primary option for renewables is hydropower. This is notably illustrated by China's National Climate Change Programme, indicating that current measures are expected to lower greenhouse gas emissions by 2010 by 500 Mt  $CO_2$  thanks to hydropower, 60 Mt  $CO_2$ thanks to wind, solar, geothermal and tidal energy and only 30 Mt  $CO_2$  thanks to bioenergy, essentially for heat and power (NDRC, 2007).

Over the past two decades, China's vehicle market has been the fastest growing in the world (+12%)each year, Latner et al., 2007). China's consumption of crude oil totalled 323 million tons in 2005, including net crude-oil imports of 119 million tons. Consistent with new car use, the annual average growth rate for gasoline and diesel consumption during the period from 1990 to 2004 reached 6.8% and 10.1%, respectively. Thus, China views biofuel as a necessary strategic component to reach independence of imported oil (Latner et al., 2006). The development of biofuels, started in the late 80s, led to the first ethanol production in 2002. In 2004, the first recorded ethanol production was 300 000 tons and it increased more than four-fold within two years to reach 1.3 million tons in 2006, and an estimated 1.45 million tons in 2007. Ethanol is primarily converted from maize (>80%). Biodiesel, which was not introduced in the development programme until 2006, is mostly produced from animal fat or waste vegetable oils.

The 11th Five-Year Plan for biofuels that had suggested an implementation plan leading to a production of 5.2 million tons of biofuels by 2010 was not approved for food security concerns (Latner et al., 2007). China is already a net importer in all the major edible vegetable oils, the largest importer in the world, and a net sugar importer (Latner et al., 2006, 2007). As ethanol already accounts for 40% of the industrial maize use, higher demand for ethanol could turn China from a net exporter of maize into a net importer (Latner et al., 2006). Therefore, the government focuses more on the use of other crops such as cassava, sorghum, and the use of feedstock grown on non-arable lands, notably cellulosic feedstock for ethanol and Jatropha for biodiesel. It has already launched an E10 mandate in nine provinces that will be expanded to some of the other thirteen provinces. It is not clear today whether the ambitious government's target of a biofuel share of 15% of total transportation fuels by 2020 [about 12

million tons of biofuel (Latner et al., 2006)] remains on the agenda, since the Plan has been rejected. A realistic target would be 3 to 4 million tons of biofuels by 2010 (Latner et al., 2007), but the Chinese government needs to draw up new policies to ensure that its biofuel targets can be achieved efficiently and economically, said a researcher within the Chinese National Development and Reform Commission (Stanway, 2008).

The European Union, the United States of America and China are major emitters of greenhouse gases, and therefore show a growing political will to reduce their emissions, notably by developing cleaner energy sources. Nevertheless, many other countries, even some minor polluters, are implementing national strategies and policies to develop biofuels playing a more or less important part within renewables development plans. This is notably the case in some South American countries such as Colombia and Peru, but also in Asian countries such as India, Japan and Thailand. Industrialised countries may emphasise more their role in greenhouse gas reduction and energy diversification, while developing countries promote biofuels especially as an opportunity to foster rural development and save foreign exchange. In all cases, biofuels are perceived as a means to contribute to energy security, when concerns are growing with the surge in oil price (Kojima and Johnson, 2005). However, the greatest barrier that has hampered biofuel large-scale commercialisation is their high cost of production compared with conventional fuels, two to four times higher (VIEWLS in Pelkmans et al., 2006). Thereby, biofuel policies notably consist of implementing economic incentives to counterbalance high production costs and make biofuels competitive.

# 4.3 Economic Incentives

Biofuels cannot be the panacea that petroleum has been for decades. The feedstock has to be produced (feedstock prices account for from two-thirds up to 90% of the total costs of 1st generation biofuels, Wiesenthal et al., 2007) and transformed, while petroleum just has to be looked for and exploited, which until resources started to become scarce, was largely cost-effective. Moreover, prices of agricultural commodities, especially those of crude materials, are highly volatile due to fluctuations in price inelastic supply and demand, meaning that a small shift in supply or demand results in a large price change. Supply may widely vary following climatic hazards and demand on such competitive markets, as agricultural commodities have only limited substitutes, and can be severely impacted by diverse factors, e.g. large purchase by governments (Clem, 1985). These fluctuations are further influenced by growing spill-over effects from one market to another, as global markets have become increasingly intertwined across all commodities and between commodities and the financial sector. Greater price uncertainty implies higher risk and growing speculations, which in turn can initiate a vicious cycle of even more enhancing price volatility. On the other hand, this uncertainty tends to limit opportunities to access credits and result in the adoption of low risk production technologies at the expense of innovation and entrepreneurship (FAO, 2007), which is notably critical in the development of second generation biofuels. Facing the high cost of biofuels, even Brazil's ethanol infrastructure model, that relies on an optimum combination of the very productive sugar cane and favourable climatic conditions, required huge taxpayer subsidies over decades before it could become viable (Xavier, 2007). Today, Brazil continues to maintain a significant tax differential between gasoline and ethanol (Kojima and Johnson, 2005).

Figure 7 gives an overview of the current and foreseen production costs of biofuels compared with petroleum products. Except for sugar cane-based ethanol and animal fat-based diesel, drastic cost reductions are still necessary for biofuels to become clearly competitive by 2030. Tax incentives, administered pricing, restrictive trade policies, credits and numerous other economic incentives are in force worldwide to underpin the production or consumption of biofuels, notably by making them artificially competitive.

Demand-side instruments, such as tax incentives and obligations (e.g. mandatory blends), are the most common mechanisms that have proven to be efficient in pushing biofuels onto the market (Wiesenthal et al., 2007). Tax incentives are tax provisions that grant special tax relief designed to encourage certain behaviours by taxpayers. Tax exemption on biofuels and higher excise taxes on fossil fuels permit one to compensate for the higher biofuel production costs (Table 3) and create or enlarge a favourable price for biofuels relative to fossil ones, providing a strong incentive for the consumer to prefer biofuels over fossil ones. Hence consumer acceptance was shown in a study of the United States General Accounting Office to be essential to the use of alternative fuels in the case studies of Brazil, Canada and New Zealand (US GAO, 2000). In the same study, it was also shown that the expected decline in ethanol use in the US, if tax exemption were eliminated, would be at least 50%.

In 2004, nine European Union member states had partly or completely detaxed biofuels: Austria, the Czech Republic, France, Germany, Italy, Lithuania, Spain, Sweden and the United Kingdom. All European Union Member States that achieved a high biofuel share had a full tax break in place and high fossil fuel tax levels. However, the reverse case does not seem true, which indicates that tax exemption is not a sufficient condition to reach a high share of biofuels (Pelkmans et al., 2006). Moreover, tax exemption implies budget losses for the governments, some 1140 million Euros in 2005 (\$ US 1419 million<sup>27</sup>) in Germany (Wiesenthal et al., 2007). These losses can be particularly critical in developing countries, where gasoline taxes are often a significant source of tax revenue more supported by high-income groups. Furthermore, tax expenditures aiming at favouring biofuels may fall under little scrutiny, while public expense on biofuels might need to be weighed against other social priorities (Kojima and Johnson, 2005).

Although tax incentives appear to be necessary to create and maintain a minimum biofuel demand, they are not sufficient to reach significant levels of biofuel consumption in most countries. In particular, ethanol tax incentives in the US, for instance, have failed in enhancing US energy security because they have not created enough usage to reduce petroleum imports and the likelihood of oil price shocks (US GAO, 2000). Thus, currently implemented schemes are mixed, i.e. some kind of tax incentives and/or obligations apply in parallel (Wiesenthal et al., 2007). In France, reductions of the interior consumption tax (Art. 265 bis A from the Duty Code 2006) by an average of  $28 \in (\$ US 35.2^{24})$ per hectolitre of biofuel blended with fossil fuels is combined with a supplementary tax (General Tax on Polluting Activities, TGAP, Art. 32 from the Finance Law 2005) on diesel and gasoline sales, which do not contain a minimum share of biofuels (Luneau and Fayet, 2007).

<sup>&</sup>lt;sup>27</sup> Mean annual foreign exchange rates from the US Board of Governors of the Federal Reserve System annual databases.



Daily wholesale gasoline and diesel prices (USD/l free on board) in 12 locations versus the crude oil price compared with the production cost ranges for biofuels without any government support schemes, both now and as anticipated by 2030.

Fig. 7 Biofuels current and prospected production costs without support schemes and petroleum products price evolution (IEA, 2007b)

		Gasoline	Biodiesel	
Countries/Community	Ethanol	(US\$ per energy-equivalent litre*)		Diesel
US	0.82	0.54	0.62	0.57
	(wheat)	(with tax)	(vegetable oils)	(with tax)
	0.45	0.38		0.37
	(maize)	(without tax)		(without tax)
Canada	0.85	0.68	0.50	0.68
	0.50	0.40		0.39
EU-15	0.88	1.32	0.68	1.29
	0.68	0.40		0.40
Poland	0.80	1.2	0.82	1.1
	0.51	0.39		0.38
Brazil	0.35	0.84	0.65	0.49
	(sugar cane)	0.39		0.38

**Table 3** Production costs of ethanol and biodiesel and prices of petroleum-based fuel in major biofuelproducing countries, 2004. Data from OECD (2006)

Notes: \* Biofuel prices accommodate differences in energy content. Ethanol is assumed to contain 0.66 the energy of 1 litre of gasoline, and biodiesel is assumed to contain 0.89 the energy of 1 litre of diesel. Note: when not specifically mentioned, the numbers follow the same order as for the US, i.e. 1st: wheat, 2nd: maize; and 1st: with tax, 2nd: without tax.

In Germany since 2007, a tax exemption system has been replaced by an obligation for fuel suppliers to provide a certain share of their total sales as biofuels. Obligations encompass mandatory blends, i.e. an obligation to add a certain % biofuels to fossil fuels, obligation to bring a certain quantity of biofuels on the market (e.g. the 10% share proposed in the European Union Directive CEC, 2008), and an obligation to bring a certain biofuel quantity on the market including a tradable renewable fuels certificate system (e.g. the Renewable Transport Fuels Obligation in the UK, the Green Power Certificates System in Flanders) (Pelkmans et al., 2006). In a broader sense, fuel standards, an authorised quota system for biofuel producers or filling station obligation (e.g. a mandate to fuel distributors to offer at least one renewable fuel in Sweden, Pelkmans et al., 2006) can also be perceived as obligation incentives, since they aim at introducing given quantities of biofuels on the market. Indeed, a major interest of obligation is the long-term visibility that they may offer to industries willing to produce biofuels, especially to those taking the risk to implement new technologies. This market prospect remains relative, though, as it depends on the governmental politics for the quota amounts (Pelkmans et al., 2006). Furthermore, in an obligation scheme, fuel suppliers will pay for the additional costs, meaning that they have an incentive to opt for the lowest cost biofuels, e.g. imported or low-blend biofuels. Fuel standards and low blends do not make any biofuel visibility possible, and obligations do not push industries to go beyond mandate targets (Pelkmans et al., 2006).

Finally, if high blends or certain technologies are to be promoted, neither obligations nor tax reductions are the appropriate instruments (Wiesenthal et al., 2007). It can be noticed that second generation biofuels will count for double compared with first generation ones, when considering US or European obligation systems, which in that case is a further incentive towards these biofuels. However, supply-side instruments may be more efficient in promoting specific biofuels. Moreover, while most of the initiatives in the biofuels field have been focusing on conversion and end-use sectors so far, there is a primordial need to support raw material producers to expand and secure feedstock supply in order to/and thereby be coherent with increasing biofuel share policies (European Commission, 2007). As an example, the E5 mandate in India was suspended in 2004, two years after its implementation, due to the lack of ethanol supply.

Supply-side instruments, mainly capital grants and feedstock support, have had limited success in pushing biofuels so far (Wiesenthal et al., 2007). Indeed, support for production facilities for 1st generation biofuels does not drastically impact the cost of biofuel due to the fact that capital does only represent a marginal part of the total cost in comparison with feedstock  $(0.01 \in \text{per litre with } 10 \text{ million} \in \text{support for a}$ 15–20 million € investment to build a typical large biodiesel plant, Pelkmans et al., 2006). However, capital grants play an important role in fostering the development of the not yet mature second generation of biofuels, firstly as an incentive to build up demonstration plants, but also given that capital costs account for much more of the total cost when compared with 1st generation biofuels, some 60% and above (JRC/EUCAR/CONCAWE, 2006). The same pattern is observed with feedstock support. Although feedstock represents a large part of the total production costs, the energy crop premium indeed only lowers 1st generation biofuel costs by 0.01 to  $0.04 \in$  per litre. But as capital grants or loan programmes, feedstock subsidies appear to be an efficient instrument in order to support special types of biofuels (Wiesenthal et al., 2007).

In Europe, agricultural subsidies are defined by the Common Agricultural Policy (CAP). In 1992 following the Mac Sharry reform, liquid biofuels appeared as a way to compensate for the set-aside obligation that aimed at tackling the issue of overproduction and corresponded to 15% of the arable crops area in 1993–1994. In exchange, farmers were entitled to compensation payments (area and headage payments), and were allowed to grow non-food crops on set-aside lands (regulation n° 1765/92). In France, rapeseed methyl ester was favoured because it permitted the cultivation of the greatest area of set-aside lands for a given amount of public financial support considering its low yield per hectare (3.3 t/ha on average in 2005) (Sourie et al., 2005).

In 2003, the Fischler reform established "an income support for farmers", the single payment scheme (SPS), which replaces area and headage payments, cutting the link between subsidies and production. As a result, farmers can respond freely to increasing demand for energy crops (CEC, 2005). In particular, crops, which were eligible for direct payments only as non-food crops grown on set-aside lands, can from now on be grown on any kind of land. Moreover, in the past, only a limited range of energy crops could benefit from support, whereas this new reform has paved the way for farmers to grow more energy crops, including short rotation coppice and other perennial crops (CEC, 2005). Non-food crops including energy crops can be grown on set-aside lands under the

condition that the use of the biomass is guaranteed by a contract between the farmer and the processing industry or by the farmer if the processing occurs on the farm (CEC, 2006a). In these cases, the 2003 reform also introduced a  $45 \in$  /ha premium for energy crops grown on non-set-aside land, the so-called Carbon Credit, with a budgetary ceiling of Maximum Guaranteed Area of 2 Million hectares subsidised in the EU (regulation1782/2003, amendment<sup>28</sup> in 2007). In 2006, this energy crop premium was already applied to some 1.2 to 1.3 Million hectares (Wiesenthal et al., 2007). In a further push to encourage the production of feedstock for renewable energy production, the Commission also proposed allowing the Member States to grant national aid of up to 50 percent of the costs of establishing multi-annual crops on areas on which an application for the energy crop aid has been made<sup>29</sup>.

Finally, in November 2005, a major reform of the sugar regime was agreed. It aims at a progressive cut in price support of 36% over four years and the reduction of EU sugar subsidised exports from the current level of 7.6 Mt to the agreed Uruguay Agreement<sup>30</sup>'s limit of 1.4 Mt (OECD/FAO, 2007); meanwhile, "partial compensation" was introduced in the form of a direct decoupled payment. Sugar for the chemical and pharmaceutical industries and for ethanol production is excluded from production quotas. The lower sugar production quotas and the lower sugar price paid to EU farmers are likely to foster the production of sugar beet for ethanol, which is also eligible for the energy crop premium. The new US Farm Bill will also emphasise the role of agricultural subsidies to foster the development of biofuel with increased support to farmers. Despite the willingness of the former US President, G. Bush, to privilege direct payments upon coupled ones<sup>31</sup> or even the recent CAP reform, feedstock subsidies still imply market distortion.

As Mrs. Corre, Director General of the European Union of Ethanol Producers (UEPA), pointed out, a balanced trading framework is a pre-requisite to ensure the viability of the nascent industry and offer win-win opportunities, especially to developing countries. Also, biofuel producers need a self-running market with long-term visibility, and a 2–3-year perspective based on a yearly-tailored fiscal budget does not offer that. Internalising the external benefits of fuel ethanol is the greatest help of all which could solve the problem (in EurObserv'ER, 2006). However, these externalities are not easy to quantify and to be given any value and the cost/benefit analysis also includes a lot of socio-economic factors that are in constant evolution, e.g. economic incentives, the cost of an oil barrel, etc. Taking the example of greenhouse gas emissions, which are given an economic value within the Emissions Trading Scheme (ETS), the assessment of CO<sub>2</sub> equivalent emissions throughout biofuel chains, first of all, still lacks precision (see part 5).

Transportation biofuels are not included in the current Emission Trading Scheme for CO<sub>2</sub> permits. The abatement cost of using biofuels to reduce greenhouse gas emissions, by substitution of fossil ones, is indeed very high due to their high production costs. Therefore, unless a given biofuel permits drastic greenhouse gas reductions with low costs (Fig. 8: Cane ethanol in Brazil, US/ton GHG reduction = US 25) the price of the ton  $CO_2$  equivalent, i.e. of the tradable permit, would be too high within the current ETS "pricing engine" (Fig. 8: Grain ethanol in IEA countries, US/ton GHG reduction = US 200-500 and biofuels would not be a favoured solution for greenhouse gas reductions compared with green heat and power if permits were the only incentive (Fulton et al., 2004). This system makes emission reductions possible wherever abatement costs are the lowest, i.e. this trading scheme cannot lead to efficient greenhouse gas reductions where they are the highest as long as production costs are not significantly reduced.

Clean Development Mechanism (CDM) projects that involve an industrialised country buying carbon credit for a developing country, which uses the payment to produce biofuels or dedicated vehicles, etc., might be a better option to foster biofuels, depending on their greenhouse gas reduction potential. Also, reduction of tariff barriers within international trade for biofuels as environmental goods (under the classification of OECD, 2003 in Fulton et al., 2004) is an interesting instrument given the wide range of biofuels' potential and production costs worldwide.

<sup>&</sup>lt;sup>28</sup> http://ec.europa.eu/agriculture/capreform/infosheets/energy\_ en.pdf.

<sup>&</sup>lt;sup>29</sup> European Commission Press releases, IP/06/1243, Brussels, 22 September 2006.

<sup>&</sup>lt;sup>30</sup> URAA, Uruguay Round Agreement on Agriculture in Marrakech, 1994.

<sup>&</sup>lt;sup>31</sup> Communication of Charles E. Hanrahan, Senior Specialist at the Library of Congress in Washington, on 01/2008 at the Agroparistech.



Fig. 8 Biofuels incremental cost per ton of greenhouse gas reduction (Fulton et al., 2004)

However, again, these opportunities are still hampered by the lack of data and agreement on real biofuels' environmental benefit.

Finally, the assessment of past biofuel success stories indicates that a portfolio of policy instruments, including supply- and demand-side instruments, is necessary to bring biofuels onto the market (Wiesenthal et al., 2007). This implies a need for interdisciplinary integration and harmonisation within ministries and governments at all levels. All over the world projected or implemented mandatory blends are on the agenda: the E10 mandate in Thailand and China, planned B2 and E10 mandates in Latin America, the possible ethanol blending mandate in Japan, the Brazilian mandatory B2 blend introduced in 2008 and to be increased to B5 in 2013, etc. However, these targets are far from being reached and illustrate a lack of policy harmonisation in terms of bioenergy strategy on a global level.

Furthermore, difference in blend standards can impact international trade as biofuel producers from different countries might not be supported by subsidies or taxes in the same way when considering various blend characteristics needed to get this help, as claimed by the European Biodiesel Board (EBB) in Bioenergy Business, 2007. This also stresses the need for harmonisation across several sectors and especially involving the automotive sector. Collaboration with car manufacturers has appeared crucial to ensure biofuel compatibility with engines, that was necessary to offer biofuel producers and consumers warranty (Wiesenthal et al., 2007). The introduction of policies on vehicle technology standards, so that all new vehicles would be compatible with a specific mixture of biodiesels, would permit one to lower the production costs of such vehicles and further incite biofuel consumption (Fulton et al., 2004). It would also pave the way to more flexibility and more coherence with other policies, which is lacking; as the example of the Spanish biofuel domestic surplus shows (Bioenergy Business, 2007).

Despite existing policies and measures, the major uncertainty factor, to decide which biofuel path should be fostered or not and to define better who should bear the additional costs, is due to diverging results on the energetic and environmental balances of biofuel chains. Calculations and results can drastically differ among studies and the lack of transparency behind hypotheses or the data quality also lead to some confusion. Therefore, growing doubts about the real ability of first generation biofuels to reduce overall greenhouse gas emissions and growing awareness of negative impacts of biofuel production on biodiversity, water and soil, point to the need for great caution in promoting biofuels further. Better knowledge of lifecycle greenhouse gas emissions from all energy uses of biomass, and strong sustainability criteria for biomass production, also addressing knock-on effects due to indirect land-use change, are needed to fully judge the benefits and limitations of biomass use (EEA, 2008).

# 5 Biofuels and Greenhouse Gases

# 5.1 Assessing the Environmental Impacts of Biofuels

#### 5.1.1 Life Cycle Assessment of Biofuel Chains

#### The Life Cycle Assessment Tool

Life cycle assessment (LCA) is a holistic methodology developed in the 1980s<sup>32</sup>, which assesses the potential environmental impacts of a product considering every step of the commodity chain from "the cradle to the grave". When comparing biofuels and fossil fuels, LCA appears to be an inescapable tool because the production of biofuels must be optimised, considering the environmental impacts throughout the whole commodity chain in order to avoid pollution trade-offs between ecological compartments or processing steps. This is, indeed, very important in "eco-design"; not to solve one environmental problem merely by shifting it to another stage in the product's life cycle (Guinée, 2002) and this is particularly crucial when it comes to decisions on national and international levels on global issues such as energy and climate change. The LCA methodology consists of 4 steps: "goal and scope definition", "inventory of extractions and emissions", "impact assessment" and "interpretation". An iterative approach should ensure that the system boundary and the inventory have been correctly adjusted, so that a comprehensive inventory of emissions makes it possible to correctly and completely characterise the selected

<sup>32</sup> By BUWAL, Bundesamt für Umwelt, Wald und Landschaft (Swiss federal office) and SETAC, Society of Environmental Toxicology and Chemistry (international scientific society). impacts. A complementary sensitivity analysis then allows eliciting the weight of input data uncertainty and model assumptions on the final LCA results. Although the LCA tool has been standardised by the ISO norms 14 040 series (40/41, 43, 47–49, 1997/98, 2000/01<sup>33</sup>), there are several methods to apprehend the impacts and their characterisation, and no ready LCA adaptable to various cases. This firstly implies that the whole analysis has to be started from zero each time, but also that results might considerably vary between studies due to diverging background assumptions. LCA has been performed a lot during the last decade, notably in order to compare the environmental performances of bioenergy chains with one another or with fossil fuel chains. In this sense LCA can serve for decision-making but under some conditions on its construction that should lead to consensus on the LCA results.

Figure 9 shows a simplified scheme of a representative system boundary for a bioenergy chain from the extraction of raw materials to the combustion of the biofuel, the so-called Well-to-Wheel (WtW) system boundary. Delineating the system boundary is a decisive step of the "goal and scope definition" step of LCA, although it might evolve through iterative analyses of result sensitivity to assumptions. There is nowadays a global consensus on this WtW system boundary, although some divergences still appear, notably concerning the accounting of energy invested in farm machinery and infrastructure capital (Farrell et al., 2006). A distinction should be observed, though, between WtW assessments, that elicit the impacts of the fuel combustion, and Well-to-Tank (WtT) assessments, that assume total fuel combustion without further impact assessment. Renewable energies, upstream in the chain, are not included in the system boundary; essentially solar energy for photosynthesis, but also the so-called indirect energies such as human work, for example.

Within the system boundaries, from the first extraction to the last emission, all elementary flows are accounted throughout every step of the "tree of the product life cycle" in accordance with a quantified functional unit, i.e. the function that the studied product

<sup>&</sup>lt;sup>33</sup> ISO 14040:2006 and ISO 14044:2006 replace the previous standards (ISO 14040:1997, ISO 14041:1999, ISO 14042:2000 and ISO 14043:2000). The new editions have been updated to improve the readability, while leaving the requirements and technical content unaffected, except for errors and inconsistencies.



Fig. 9 Biofuel LCA simplified well-to-wheel system. LCA: life cycle analysis

is to fulfil, e.g. to provide 1 MJ. The "tree of the product life cycle" corresponds to the succession of the unit-processes. The blue boxes represent the main classes of processes within bioenergy chains, each one encompassing several unit-processes. For each unitprocess, data are recorded on the inputs, the emissions, waste flows, and other environmental exchanges that are typically assumed to be linearly related to one of the product flows of the unit-process. All unit-process are linked through intermediate product flows, which makes the process system model linear with respect to the quantity of function it provides (Rebitzer et al., 2004). Inputs include raw materials and energy. Outputs are the products, the work or energy output and the polluting emissions. "Products" include the product of interest and all co-products; i.e. all the products, included biomaterials, waste or energy, that are concomitantly produced although all processes aim at optimising the production yield of the one product of interest.

A distinction can be made between co-produced materials that are directly generated from part of the feedstock such as straw or meal, and by-products such as glycerine or heat that are sub-products of other production processes (Malça and Freire, 2006). In this review all these "secondary" products are included in the "co-products". Direct emissions are emissions

occurring during the production of the biomass due to natural biochemical and physical mechanisms within the soil-plant-atmosphere ecosystem. Indirect emissions can also occur on a wider space and time scale, following further reactions affecting the substances previously emitted, or upstream in the chain due to land-use changes. Indirect emissions also encompass methane emissions through fodder digestion by livestock. The term indirect emissions will be, in this review, reserved for these "natural second-order" emissions, whereas emissions through cultural operations, transport, conversion processes, storage, etc. will be referred to as industrial emissions (Fig. 10).

Energy ratios are a critical aspect of bioenergy chain assessments, since an important matter is to determine the fossil primary energy savings. Nevertheless, there is still a lack of consensus concerning the definition and designation of energy efficiency indicators to be used in a life-cycle perspective in particular to characterise renewable energy systems (Malça and Freire, 2006). The respective definition and use of among others "energy efficiency", "overall energy balance", "gross/net energy requirement" and "energy renewability efficiency" (Malça and Freire, 2006) will also be clarified when comparing bioenergy chain assessments.



Fig. 10 Natural and industrial emissions of a bioenergy chain (without combustion)

#### Biofuels versus fossil fuels

Advantages of biofuels over fossil fuels depend on the environmental impacts that are considered. If e.g., savings of fossil resources as well as greenhouse gases are given the highest ecological importance, all biofuels compare favourably with their fossil counterparts if competition for other uses of the resource is not considered (Quirin et al., 2004), whereas taking into account the impacts such as acidification, eutrophication or ozone depletion reverses this trend. Indeed, these drawbacks are linked to intensive agricultural production that notably leads to nitrogen compound emissions responsible for those bad impacts on the environment. The relative advantages of biofuel chains between one another depend on the feedstock. Figure 11 illustrates the results of a comparative study of 109 biofuel chains by the IFEU, i.e. the ranges of greenhouse gas and primary fossil energy savings by substituting fossil fuels with the biofuels assessed in some 64 studies (Quirin et al., 2004).

Comparison on the basis of a unit MJ energy content of biofuels (Fig. 11a) makes it possible to compare all the chains, including biofuels from residues, whereas the comparison on a hectare basis (Fig. 11b) introduces a land-use perspective. Amongst 1st generation biofuels, bioethanol from sugar cane, molasses and sugar beet, as well as biogas from wastes (here compared with gasoline, but also advantageous compared with natural gas) show the best combined performances in terms of both greenhouse gas and primary fossil energy savings (Fig. 11a). Greenhouse gas emissions are partly linked to the combustion of fossil primary energy input throughout the chain; both savings are therefore connected and their ranges appear within similar orders of magnitude.

Due to the high energy content of oilseeds and less complex processes, relatively less primary energy input is necessary to produce a MJ of biodiesel or pure vegetable oil compared with a MJ of ethanol, whose distillation and dehydration are energyintensive. In comparison, the primary energy input in refining gasoline or diesel is not significantly different between the two fuels. Therefore, diesel substitutes can save the most fossil energy. To a lesser extent, they also mostly emit less greenhouse gases than ethanol from starch crops and ETBE. Greenhouse gas savings vary a lot among biofuel chains and biodiesel roughly performs better than the worst ethanol chains but is worse than the best ethanol candidates. Ranges in energy and greenhouse gas savings of biodiesel chains are less connected than those of ethanol, since the conversion processes consume less energy. Greenhouse gas emissions depend therefore more on the feedstock types and the cropping systems. In the case of sugar crops, energy inputs are partly compensated for by higher yields per unit of agricultural input.

Further savings are also due to the co-products. This is notably the case with sugar cane, whose stalk or bagasse are burnt to produce energy input, or in the case of rapeseed biodiesel compared with pure rapeseed oil. The latter is not refined and would by itself lead to higher saved fossil energy and greenhouse gases. However, the production of biodiesel coproduces glycerine that substitutes fossil-based glycerine. This co-product can be used as a substitute for chemical glycerine or as animal feed. The savings from displacing this otherwise produced glycerine are included within the system of the biodiesel chain, and can be particularly significant in the case of substituted chemical glycerine, whose manufacturing is very energy-intensive (JRC/EUCAR/CONCAWE,



EtOH wheat EtOH com EtOH potatoer ETBE sugar be ETBE wheat ETBE pota ethanol lignoc **Biodiesel sunfowers** Biodiesel rapes Biodiesel canola \*\*\* **Biodiesel** coconuts Biodiesel soy bean etable oil ra Vegetable oil sunfl Primary energy Biogas cultivated bio Greenhouse effect -15 -10 -20

GJ saved PE/ha\*a

-100

-50

0

EtOH sugar can

EtOH sugar beets

EtOH lignocellulo

-150

-200

**Fig. 11** Results of the energy and greenhouse gas balances of the analysed biofuels as compared to their fossil counterparts (gasoline for ethanol, biomethanol, biogas and  $H_2$ ; MTBE for ETBE; diesel for biodiesels, vegetable oils, DME and BtL) in MJ or GJ saved primary energy and saved metric tons CO<sub>2</sub> equivalent per MJ of biofuel (**a**) and per hectare a year (**b**). Negative values imply advantages for the biofuels; the zero mark means that the CO<sub>2</sub> equivalent emissions are balanced when the total

2006). Nevertheless, these savings are no longer true as soon as the glycerine market is satisfied. There are many diverse uses of glycerine, even as feedstock for energy (IEA, 2007b), but there are also many industrial processes that lead to glycerine co-production (Russi, 2008). Assuming that large-scale biodiesel production will cause a saturation of the glycerine market (Larson, 2006), some authors do not consider any savings from glycerine co-production (Russi, 2008), which equal using economic allocations for co-product handling.

From a hectare perspective (Fig. 11b), ethanol from sugar crops performs better than ethanol from starch crops, and much better than the bio-substitutes of diesel, due to higher crop yields. ETBE makes more significant savings possible than ethanol because of its high energy-related hectare yield that permits one to displace higher amounts of fossil fuel; for each MJ of ethanol that replaces gasoline, 3 MJ of ETBE<sup>34</sup> replace

life cycle (biofuel minus fossil fuel) is considered (Quirin et al., 2004)).

\* The spectrums for ethanol from lignocellulose are not unrestrictedly comparable with the others, since lignocellulose from biomass and that from organic residues are put together here.

\*\* Only from cultivated biomass

\*\*\* Canadian brand name of summer rapeseed

t saved CO-equiv./ha\*a

MTBE. However, on an energy-unit basis ethanol is slightly more interesting than ETBE, as ETBE requires a further processing step compared with ethanol. ETBE is, though, here compared with MTBE, which is produced with more energy input than gasoline but emits a little less greenhouse gas, as the bulk of the energy is provided by natural gas for MTBE rather than heavier hydrocarbons in the case of gasoline (JRC/EUCAR/CONCAWE, 2006).

When comparing gasoline blended with MTBE, ETBE or ethanol (50% ex-wheat, 50% ex-sugar beet) combined with isooctane that compensates for the ethanol higher density, ETBE permits more  $CO_2$ equivalent reduction per MJ of ethanol (Croezen et al., 2007; Higgins, 2007). Blending in ethanol or ETBE reduces the petroleum base fuel requirement for butanes and reformate octane numbers. Compared with a LCA in which ethanol and ETBE simply replace MTBE, these modifications in the refinery operations cause an additional 3% GHG savings in the case of ethanol, and about 20% in the case of ETBE. In the case of

<sup>&</sup>lt;sup>34</sup> 3 MJ of ETBE are produced from "1 MJ of ethanol and 2 MJ of isobutylene".

ethanol, its high vapour pressure requires a counterpart reduction of the petroleum base fuels' vapour pressure that partly offsets the savings due to the ethanol's higher octane numbers. In the case of ETBE, less volume of petroleum base fuels is needed for the same volume of ethanol (5%) with an even lower octane number requirement. Although this advantage for ETBE is to some extent undone by the higher GHG emissions related to its production and that of extra isobutylene, ETBE finally results in a higher net GHG reduction of 61 kg CO<sub>2</sub>-eq/GJ ethanol converted into ETBE, against 37 kg CO<sub>2</sub>-eq/GJ ethanol when 5 vol% ethanol is added in pure form (Croezen et al., 2007). Overall more greenhouse gas reduction is possible though, when engines are modified to run with higher blends of pure ethanol.

Results on 2nd generation biofuels are still theoretical and provisional, since their production is still at a pilot stage and technology keeps evolving, which is notably illustrated by the wide spectrum of lignocellulose or H<sub>2</sub> biofuel chains (Fig. 11a). Moreover, 2nd generation biofuels can proceed from a wider range of feedstock, diverse organic residues and theoretically all possible biomass. Regarding the spectrum of possible pathways, 2nd generation biofuel assessments are still scarce.

Mature 2nd generation biofuels are expected to have less impact on the environment. This firstly lies in the fact that the whole plant is transformed, increasing the energy yield for the same mechanical and chemical treatment, which implies that the environmental load is lower for the same energy unit. Furthermore, in the case of perennial grasses, the environmental advantages could be even more consistent. The biomass of perennial grasses has higher lignin and cellulose contents than the biomass of annual crops (Lewandowski et al., 2003). Perennial energy crops and short rotation forestry or coppice (SRF/SRC) generally have less impact on: soil erosion, nutrient inputs into the ground and surface water, pesticide pollution, and water abstraction. In contrast to annual crops, perennials require only one cultivation activity, i.e. preparation for planting, over a 10- to 20-year duration, and minimal nitrogen inputs (Heaton et al., 2004). Typical energy input/output ratios vary between 1/10 and 1/20 (IEA, 2006).

Perennials also have more extensive root systems present throughout the year, thus providing increased resistance to soil erosion and a more effective means of trapping nutrients and preventing nitrogen loss to drainage water. Production and turnover of belowground storage organs can furthermore add organic matter and carbon to the soil (Heaton et al., 2004). In addition, most nutrients also remain on the land under SRC or with perennial grasses, when the harvest takes place after the nutrient-rich leaves have dropped. As a result, soil carbon and quality tends to increase over time, especially when compared with conventional farming (IEA, 2006). Finally, annual crops on average need better quality land than perennials to achieve good productivities, whereas perennials can be grown on marginal lands, thereby achieving other potential benefits such as soil quality improvement or in some cases adding to biodiversity (Lewandowski et al., 2003; EEA, 2006b). However, perennial grasses and SRC, like other crops, may also contribute to ozone depletion, acidification and eutrophication if they are fertilised.

#### 5.1.2 Limits of the LCA Tool

#### Data Quality

When compiling results from diverse biofuel LCAs, it appears that results widely differ between studies. This variation can be explained following two intertwined tracks: the intrinsic limits of LCA methodology and the lack of scientific background knowledge. LCA was first established for industrial production; so that differences between industrial and agricultural systems originate many methodological problems for agricultural LCA. The fact that industrial systems are mostly independent of their local environment has led to a site-independent methodology for LCA. However, the life cycle steps in close contact with the environment (such as agriculture or land filling) are site-dependent by nature (Kodera, 2007). Thus, in essence, LCA implies a site-independent model of a biofuel chain at a given period of time, which is merely compatible with the dual agricultural-industrial nature of biofuel chains. The first consequence is the difficulty of collecting data sets of representative quality; i.e. relevant, transparent, precise, complete and reproducible, while data collection and compilation are already often the most work- and time-consuming steps in LCA (Rebitzer et al., 2004). On the one hand, agricultural data sets are time- and site- (soil-climate) dependent, which

implies uncertainty in extrapolation and modelling, and further variability in biofuel chain assessments. On the other hand, industrial data sets are not systematically accessible and transparent for outsider assessors.

Now, the attempt to make an exhaustive inventory should not be at the cost of data quality. According to Delucchi (2004), using literature-review estimates with an aggregated approach of processes instead of primary data from an appropriated input/output flow model for each process can amount to a percent or two of direct fuel-cycle GHG emissions. Three further percentage points in the fuel-cycle analysis would be related to data uncertainty with regard to estimates of the energy intensity of fuel production and the energy efficiency of motor vehicles (Delucchi, 2004). Comparing the life-cycle inventory for refinery products among several databases, Jimenez-Gonzalez and Overcash (2000) have shown that the variability in estimated emissions to the atmosphere, and waterborne and solid waste are approximately 50-150%, 1000% and 30%, respectively (in Lo et al., 2005). Also, models based on international or national average values might not be useful or adaptable to more specific local production conditions for decision-making on regional levels. Moreover, a LCA steady-state model cannot take directly into account the variation in market demand or the technological advances, although they can be introduced through economic allocation or as prospective assumptions. Both of these factors are crucial for bioenergy; likewise social aspects, which are also not encompassed in LCA.

The varying quality of input data makes the comparison of diverse scenario outputs more difficult. The lack of transparency and homogeneity in background assumptions between different biofuel chain assessments may hide the fact that data might not always be reliable. To deal with data quality and uncertainties, tools exist and could, if systematically associated with the results, enlighten comparisons between assessments. A "pedigree matrix" permits one to establish data quality indicators (DQIs) that give scores to data sets (1 to 5) in function of their reliability, their completeness, and their temporal, geographical and further technological correlations linked to the goal and scope of the study (Weidema and Wesnaes, 1996). These scores make it possible to distinguish processes and flows for which input data quality is poor, and to focus on these inventory parts to compare their impacts on output data among different assessments. This qualitative approach is to be completed with statistical indicators, such as coefficient of variation, that highlight the data uncertainty: i.e. the basic uncertainty linked to typical measurement errors or normal fluctuation of the variables, and an "additional uncertainty" related to the data not being of the optimal quality as reflected by its pedigree scores. This additional uncertainty can be calculated or estimated. "Default uncertainty matrices" for different specific types of data or domains could also serve as references, when the uncertainties cannot be directly assessed (Weidema and Wesnaes, 1996). Such tools might often appear to be necessary in the case of agricultural systems for which data such as probabilistic distributions and the correlations of key parameters are particularly rarely available Basset-Mens et al. (2006a).

# **Co-product Handling**

The complexity of bioenergy chains evidences limits of LCA. Indeed, LCA methodology leaves some degree of liberty, when it comes to dealing with the handling of co-products and the complexity of a wide range of environmental impacts. The LCA methodology stipulates that the product and co-products should be handled separately as long as possible, through sub-division of processes, in order to avoid problems of burden allocation. This is, however, impossible in the case of bioenergy chains, where product and coproducts come from the same feedstock and are chemically linked. In this case, methods of co-product handling are suggested but none is mandatory.

The expansion of the system boundary implies that the process leading to the production of a coproduct is taken into account as a co-function of the system, either by additive substitution or by subtractive substitution. In both cases, the flows corresponding to the production of the co-product are taken into account in accordance with energy and environmental loads of the co-product, when produced through a fossil chain; either as supplementary loads in the reference scenario (additive substitution), or as loads to be subtracted from the bioenergy chain (subtractive substitution).

The ISO 14 041 (1998) stipulates, where allocation cannot be avoided, that the inputs and outputs of the system should be partitioned between its different products or functions in a way which reflects the underlying physical relationships between them (physical allocation or apportioning). Physical allocation implies that environmental loads and energetic costs are partitioned between the product and the diverse co-products in accordance with defined mass or energy ratios. Although mass ratios are easily measurable and therefore more frequently reproducible across studies, they do not reflect a fair share in burdens. Indeed, since the biofuel justifies all the energy and material expenses, it should support the main share of the total burdens implied by its production, which is not the case, for instance, when a co-product weighs more than the biofuel itself. Energy allocation credits may be closer to the logics of the functional unit of biofuels; however, differing calculation on an energy content or energy consumed basis can also introduce further variations among studies. When physical relationships cannot be distinguished, then a financial allocation is the remaining option and consists of defining ratios in function of the market value of each product. This last option can be quite relevant in the sense that economics will still be the underlying driving factor of a biofuel chain development, while LCA does not in itself take into account economic factors. But it also may create a bias if market values are punctually considered. An economic analysis is necessary to speculate on the price evolution of the diverse products.

As explicitly recommended by the ISO norms, expansion of the system boundary is the first solution that should be examined in order to address the issue of co-product handling. The introduction of the substitutes within the system boundary permits one to elicit the real impact of the co-product production in a specific context. For instance, Prieur and Bouvart (2006) showed with the example of BtL from wood, that the relative greenhouse gas savings compared with the fossil chain could vary between around 68% and 104% when considering a substitute co-produced for electricity either from a French mix (mainly from nuclear power) or average European mix (more from coal), respectively. Variability in this case is even wider between the two types of substitute than between one of the two substitutions and mass or energy allocations.

Biofuel chains are not equally sensitive to coproduct handling. Comparing the diverse co-product handling scenarios (three allocation ratios: mass, energy or market value; and expanding the system), the burden of primary energy that comes to ethanol varies from 89.7-95.6% of total primary energy in the case of sugar beet, whereas it varies from 42.7-91.2% in the case of wheat (Malça and Freire, 2006). Kim and Dale (2002) also found that burdens of corn ethanol widely vary following the co-product handling, within the comparable range of 48-80% for both dry and wet processes. In these studies, mass allocation tends to lower the biofuel burden, whereas system expansion maximises it. Methodological choices (e.g. system boundaries and allocation methods) have a large influence which may very well override many other types of uncertainty (Björklund, 2002 in Malça and Freire, 2006). Following a pragmatic approach, it may be very relevant to introduce substitute products that are representative for the specific local production at stake. However, system expansion requires that an alternative way of generating the exported function exists and that data are available and collected. Such ways may not be found, or on the contrary, some coproducts can possibly substitute a wide range of products (Malça and Freire, 2006; Kodera, 2007).

The lack of a clear standard for choosing substitution products may lead to arbitrary choice and inconsistent calculations and results even if the methodology remains unchanged (Kodera, 2007). Ink has been spent on looking for a consensus on this methodological issue, but there is no single procedure to deal with the diverse co-products, not even for each biofuel chain. The ISO norm (14 041/1998) states that when an allocation issue arises, a sensitivity analysis on this allocation parameter should be done. Hence, each biofuel LCA should explore a range of possibilities. Scenarios of allocation ratios and substitution means should be elaborated and compared, taking into consideration the most relevant co-products substituting chains on the local scale.

#### Impact Characterisation

A lot of published LCAs for biofuel are not in fact LCAs stricto sensu. Indeed, they may be assessments based on a life-cycle approach and guided by the LCA ISO norms, but rarely assess all the potential environmental impacts (Quirin et al., 2004; Blottnitz and Curran, 2007). By its holistic nature, LCA would require assessing all the potential impacts linked to the flows inventoried; nevertheless, the attempt at an exhaustive impact characterisation also faces

methodological as well as scientific constraints. Life cycle impact assessment (LCIA) is the step that actually permits one to quantify the potential environmental impacts. It consists of classifying and aggregating the results of the inventory into category indicator results (Eq. (1)) that characterise how the environmental mechanisms for the chosen impact categories are modelled, and what the contributions are of the involved flows to these impacts.

Impact Category Indicator<sub>i</sub> =  $\Sigma_i$  [Characterisation

Factors  $(_{ij})$  × Emission/Extraction Inventory  $(_{ij})$  (1)

where "j" denotes the substances or resources (Brentrup et al., 2004).

Traditionally in LCIA, characterisation factors linearly express the contribution of a mass unit of an emitted substance to a given impact category. Depending on the scope and goal of the assessment, different impact assessment methods or approaches can be used, while the inventory remains the same. They actually differ in the choice of impact categories, the logic and approaches underpinning the characterisation (models and indicators). The main discussion here lies in the consideration of either midpoint impacts, or endpoint impacts, i.e. damage-oriented assessments.

Midpoints are considered to be points in the causeeffect chain (environmental mechanism), between stressors and endpoints (Bare et al., 2000 in UNEP, 2003). Midpoint impact categories are, for instance, climate change, stratospheric ozone depletion, human toxicity, acidification, eutrophication, etc. In damageoriented assessment, impact pathways connect the inventory results across midpoints to one or more of the damage categories; i.e. classes of endpoints defined as areas of protection (AoP) that can be impacted both in their intrinsic and functional values (Jolliet et al., 2004). There exist different classifications of AoP that are actually connected; e.g. human health, the natural environment and manmade environment, human health, biodiversity (or ecosystem health), and natural resources (Heijungs et al., 2003). Life support functions (LSF), climate regulation, hydrological cycles, soil fertility and biogeochemical cycles, are classes of midpoints, just like areas of protection are classes of endpoints (Heijungs et al., 2003). Depending on the state-of-the-art knowledge, the representation of a pathway link may vary from a fully quantitative

description, involving new contribution indicators such as disability-adjusted life years (DALYs), for instance, to a short qualitative description of the expected causal impact on subsequent pathway links (Jolliet et al., 2004).

Some LCA practitioners argue that endpoints are the elements of an environmental mechanism that are in themselves of value to society (Udo de Haes and Lindeijer, 2001 in UNEP, 2003). Others fear that uncertainties, due to a lack of sufficient data and robust models, may be extremely high beyond midpoints (UNEP, 2003). Hence midpoint approaches, such as CML (Heijungs et al., 1992), Eco-indicator 95 (Goedkoop, 1995) and EDIP (Wenzel et al., 1997), have been the main widespread approaches historically, whereas endpoint ones, such as EPS (Steen, 1999) and Eco-indicator 99 (Goedkoop and Spriensma, 2000) are more recently gaining interest, since they imply the analysis of trade-offs between and/or aggregation across impact categories. They also permit an integrative estimate of environmental externalities by monetary valuation of welfare losses due to impact on the AoP, which are fundamentally linked to societal values. Endpoint LCA results can be interpreted in light of marginal impact costs or distance-to-target performances, and can in this sense clearly serve to help design market-based internalisation instruments (e.g. taxes) (UNEP, 2003). Endpoint assessments are hence more directly understandable and useful for decision-makers; however, they lose in transparency when weighting is required to compare across categories, which does not systematically elicit links between midpoints and endpoints.

Therefore, both midpoint and endpoint approaches provide useful information with a trade-off between reliability and relevancy, and should be, to some practitioners' mind, conducted in parallel to determine how the results are affected (UNEP, 2003). In the case of biofuel LCAs, midpoint assessment appears to be relevant enough, though, given that impacts in terms of "climate change" and "fossil resources" are of central importance, and that uncertainties linked to data quality and co-products may already be a significant source of misinterpretation. Furthermore, a biofuel LCA is to be compared with a fossil fuel LCA, while weighting<sup>35</sup> across cat-

<sup>&</sup>lt;sup>35</sup> Weighting and normalisation are two non-mandatory steps in LCA methodology.

egories or other subjective scoring is not suitable when it comes to comparative assertions (ISO 14 042).

A more crucial issue is to improve impact characterisation, notably in order to account for the location and time of the emissions, waste generated, and resources depleted better, as well as the geographical zone and time period over which the contributions to different impacts should be considered (Pennington et al., 2004). Linear characterisation (Eq. (1)) aggregates the environmental loads at the time of the assessment, and neither takes into account the substance background concentration nor its temporal and geographical dependency on exposure and fate (UNEP, 2003). This implies that all impacts, irrespective of the moment and the place that they occur, are equally included (Udo de Haes et al., 1999). The fundament, referred to as the "less is better" principle, is that a pollutant remains a pollutant even if it is emitted in a place where it will not cause any harm; as such its emission shall be considered as contributing to potential impacts independently of site and time (Heijungs et al., 2003). It follows that impact categories are assumed to be independent of one another, and unless a precise scientific background permits one to justify the hypothesis to partition the contribution of the same substance to several impacts, the substance flow shall contribute in its entirety to each impact (Guinée, 2002). Firstly, these assumptions only lead to calculating potential impact scores, not actual damage (Khalifa, 1999; UNEP, 2003). Potential impacts thus represent the worst-case scenario, where some redundancy of a substance contribution is preferred to the risk of not considering its contribution to one impact. However, practitioners should be aware that double-counting may lead to poor decisions and that their models should try partitioning burdens as far as the state-of-the-art knowledge on the causality chain permits it (Reap et al., 2008).

Secondly, this dose-response modelling is not sufficient to describe complex environmental mechanisms, especially those where thresholds intervene (Khalifa, 1999; UNEP, 2003; Pennington et al., 2004). Simplifying assumptions and available scientific knowledge influences the accuracy of the indicators, which may vary among impact categories due to discrepancies between models and the corresponding environmental mechanisms. Hence the applicability of the characterisation factors depends on the accuracy, validity and characteristics of the models used (UNEP, 2003; Basset-Mens et al., 2006a). Lack of knowledge on the dose-response determinisms may jeopardise the reliability of the impact assessment. This is particularly true in the case of impacts on biodiversity or natural habitats, for example, for which the mechanisms are especially complex, and may also include other determinisms than chemical or physical ones. Khalifa (1999) emphasises that impact assessment within LCA is notably lacking in reliability for the impact categories of eutrophication, photochemical ozone, ecotoxicity, loss of habitats and biodiversity, as existing thresholds and non-linear dose-response notably are not considered. The reliability of the assessment is also in essence lower when it comes to local-range impacts that are more specifically dependent on the local ecosystems than global impacts. Truncations and assumptions about global homogeneity and steady-state conditions introduce the most severe errors in impact assessment. Indeed, first setting arbitrary time horizons skews results in favour of short- or long-term impacts, thus ignoring spatial variation, local uniqueness and environmental dynamics discounts the influence of environmental stress concentrations, leading to inaccurate estimates of potential damages (Reap et al., 2008). Spatial information in LCA is actually mandatory in order to contribute to solving the poor accordance between potential impact as calculated in LCA and the expected occurrence of actual impact (Khalifa, 1999; Heijungs et al., 2003; Potting and Hauschild, 2005).

Sophisticated LCIA have been developed in order to improve the level of detail, regarding in particular the temporal and spatial dimensions of the impacts. Nevertheless, only a few integrated approaches have been proposed so far (UNEP, 2003) and they were essentially developed within an endpoint characterisation, whereas regionally differentiated midpoints would also be better indicators (Heijungs et al., 2003). In particular, models have been used to determine regionalised fate and exposure factors, in order to account for background load and a priori tolerance of ecosystems to the emissions (Potting and Hauschild, 2005). For instance, Potting et al. (1998) used the RAINS model (IIASA) to produce acidification factors to be used within LCA in order to simulate acidification discrepancy better across 44 regions in Europe. This model integrates information on emission levels for each region with information on long-range atmospheric

transport in order to estimate patterns of depositions and concentrations for comparison with critical loads and thresholds for acidification, eutrophication and photochemical ozone creation. These critical load functions (weighted for the size of the ecosystems) are used to construct so-called "protection isolines" for the grid element that consist of all combinations of S and N deposition for which a given fraction of ecosystems does not exceed critical loads, and thus in RAINS terminology is assumed to be protected against the adverse effects of acidification. Regional acidification factors were calculated by reducing one by one the emission levels of each separate region by 10%, and then relating the result to the reference situation (the initial emission level and area of unprotected ecosystems). Hence acidification factor (AFs,i in ha/tons) directly relates a change in emission of substance (s) in a region (i) to the change in unprotected ecosystems in its total deposition areas. A similar approach was notably used to calculate regional factors for terrestrial eutrophication (with  $NH_3$ ,  $NO_x$ ).

Impact characterisations are integrated over an infinite period of time, since the variation in emissions on the regional scale during the time period for integration is considered as marginal when compared with the total contributing emissions (Potting and Hauschild, 2005). Indeed, LCA steady-state assumption is founded on the "multiple sources-multiple receptors" character of present environmental problems, i.e. the temporal variation in the contribution from a single source emission is usually to a large extent cancelled out against the high background exposure from all sources together. Moreover, the large impact area of an emission and the overlapping with impact areas of neighbouring sources make the precise location of a source of less importance, which makes it possible to determine site-dependent factors on a regional scale (Potting and Hauschild, 2005). However, these approximations will not be true for local (exposures within the first kilometres from the source) or time-dependent impacts (such as very slow emissions or synergic impacts) (Potting and Hauschild, 2005). Modelling the combined impacts of agricultural inputs, the climate and the hydrological functioning of catchments, Basset-Mens et al. (2006a) determined N apparent fate factors that describe the part of leachable N that actually contributes to the annual stream nitrate flux. This study emphasises that, on

the regional scale, these nitrate fate factors can imply large variations compared with the results of a standard LCA methodology. In the case of pig production, the eutrophication result was reduced by 5% to 32%, and the climate change impact varied between "no change" and an increase of 200% (Basset-Mens et al., 2006b).

Temporal and spatial dimensions are tightly intricate together. Especially meteorological conditions influence the determination of fate and exposure factors. Emission timing at different rates and locations defines site-specific emitting, fate and exposure conditions. For instance, the acidification factors calculated for the reference years 1990 and 2010 show that the difference between different calendar times can be notable (Potting and Hauschild, 2005). This would also be particularly relevant in the case of the region-specific fate factors for airborne nitrogen compounds causing aquatic eutrophication by Huijbregts and Seppälä (2000). These region-specific fate factors were modelled given European emission and meteorological data from 1985 to 1995. Regional NH<sub>3</sub> and NO<sub>x</sub> fate factors express the fractions of these airborne emissions that actually end up in the aquatic environment, taking into account both direct deposition in the freshwater and marine environments and run-off from terrestrial systems into the aquatic environment. Now, for a short time frame, these regional fate factors may, in essence, vary depending on punctual variations in precipitation patterns compared with those used to model the fate factors. The characterisation factors used to assess the impact from a given process should also relate to the calendar time in which that process takes place (Potting and Hauschild, 2005).

However, time-dependent environmental processes may necessitate time horizons for impact integration that are not in accordance with an optimum for those site-dependent factors. Thus, time-dependent factors add to the continuing discussion within the LCA community on selecting integral limits and valuing impacts distributed in time (Reap et al., 2008). Furthermore, depending on the impact, the extraction/emission region might not be the same as the region where impact/damage occur. For both time and space, it would be necessary to distinguish factors at the extraction/emission point and impact/damage point (Heijungs et al., 2003). Thus, scale precision for spatial and temporal discrepancies might remain constrained by a geographical scale large enough to cover most of the impacts from an emission source. Site-dependent factors already mostly encompass characteristics that are relevant at a country-based level. Now, these factors appear to vary also on the regional scale. In the life cycle region-specific assessment method proposed by Yi et al. (2007), where "affected regions" are distinguished from "emitting regions", regional damage factors considerably vary within the 9 defined areas and are 0 to 3 times higher than the national average.

To conclude, LCA is a powerful tool but there remain theoretical and methodological open questions that can lead to diverging results and interpretations, in particular in the biofuel chain cases. While some practitioners work on sophisticating LCA (UNEP, 2003), some others look at ways to simplify it (Rebitzer et al., 2004). On the one hand, some limits (methodological, scope of the impacts: site independency, etc.) should be overcome to make LCA a better tool for decision-making purposes; on the other hand, it already is a challenging exercise (lots of data needed, various sources of uncertainty, crucial lack of scientific knowledge for impacts, etc.). A way to solve this puzzle may be to produce local specific LCAs, based on local inventories and a better knowledge of the local receptive environment. Although the requirement of additive site-dependent data is often put forward as an objection against spatial differentiation in LCA (UNEP, 2003; Potting and Hauschild, 2005), local data sets and fitting models with local receptor parameters can make it possible to diminish the uncertainty of the results.

Uncertainty affects all the assessment steps: from the input data uncertainty, through model uncertainty, up to uncertainty on the chosen uncertainty formalisms (e.g. determinations of probability distributions, etc.) and may, if completely represented, even entail a broad interval of imprecision that could finally make the results of comparative LCAs indistinguishable (Reap et al., 2008). As illustrated in a review by Boer (2003), comparison between LCAs of diverse commodity chains across different case studies is hampered by the lack of international harmonisation on LCA methods among the studies, whereas a within-casestudy comparison of diverse chains using the same LCA methodology appears suitable to track down the main differences in potential environmental impacts. Biofuel chains should be hence compared on a regional case-study-based scale, and further extrapolation across regions should be limited to LCAs that use harmonised methods.

# 5.2 Focus on Greenhouse Gas Emissions from Agriculture

"Climate change" is often presented as an example of well-established midpoint impact characterisation, notably because its global dimension particularly fits the "multiple sources-multiple receptors" background assumption. However, when assessing biofuel chains, it appears that greenhouse gas emissions in the field can show highly variable temporal and spatial patterns, and that these emissions contribute a significant share of the total greenhouse gas balance of the whole chain. Emissions during agricultural production contribute, for instance, 34%–44% to the greenhouse gas balance of corn ethanol in the US (Farrell et al., 2006) and up to more than 80% in the case of pure vegetable oils (ADEME/DIREM, 2002). Focusing on the agricultural phase only, many factors imply variations among biofuel chains. They are linked to the ecosystem characteristics and the cropping systems, both dependent on local conditions. After briefly describing the greenhouse gas emissions linked to agricultural activities, two aspects of the current limits in the assessment of biofuels' greenhouse gas balance will be detailed, i.e. the determinism of N2O emissions and CO2 emissions due to land-use change.

# 5.2.1 Overview of Greenhouse Gas Emissions from Agriculture

Agricultural activities contaminate the environment through three main impact pathways: land-use changes, the use of farm machines, and the use of inputs, e.g. fertilisers, that are sources of many diverse pollutants throughout their life cycle. Industrial emissions concerning the agricultural phase in a biofuel LCA encompass all the emissions linked to the production and use of machines for agricultural operations and of inputs such as fertilisers or pesticides. Briefly, the more intensive the use of machines and inputs, the higher the overall emissions of pollutants. In Farrell et al. (2006), agricultural practices across six compared corn with ethanol LCAs are responsible for 45% to 80% of all petroleum inputs and related emissions. This illustrates how the intensity of cropping systems influences the overall chain performance, but also that the lack of transparency in primary energy inputs adds to the global confusion when comparing

LCAs. More attention is paid here to emissions that are the results of natural reactions within the coupled biogeochemical cycles of C and N. In general, CO<sub>2</sub>, N<sub>2</sub>O and CH<sub>4</sub> are by-products of the microbial activity, which is characterised by a transfer of electrons, hence depending on soil redox potential, dissolved organic carbon content, and the concentrations of the relevant electron acceptors (Li, 2007). These reduction-oxidation reactions are influenced by both the natural conditions and the agricultural activities, meaning that resulting emissions can vary widely and are therefore hard to predict.

Focusing on greenhouse gas emissions, in 2005 direct CH<sub>4</sub> and N<sub>2</sub>O emissions from the agriculture sector worldwide accounted for about 5.1–6.1 GtCO<sub>2eq</sub> yr<sup>-1</sup>, equivalent to 10–12% of the total anthropogenic emissions of greenhouse gases (Smith et al., 2007). This includes 3.3 GtCO<sub>2eq</sub> yr<sup>-1</sup> of CH<sub>4</sub> (50% of total CH<sub>4</sub> anthropogenic emissions<sup>36</sup>) essentially due to enteric fermentation from livestock (27% of agricultural greenhouse gases in Baumert et al., 2005 of a total of 6.2 GtCO<sub>2eq</sub> yr<sup>-1</sup> in 2000), to rice cultivation on wetlands (10% of agricultural greenhouse gases in Baumert et al., 2005, and to manure management (7% of agricultural greenhouse gases in Baumert et al., 2005), and 2.8 GtCO<sub>2eq</sub> yr<sup>-1</sup> of N<sub>2</sub>O (60% of total N<sub>2</sub>O anthropogenic emissions) (Smith et al., 2007) produced by microorganisms in the soils (40% of agricultural greenhouse gases in Baumert et al., 2005). Annual  $CO_2$  emissions by agricultural lands are low compared with overall anthropogenic CO<sub>2</sub> emissions. The net flux between the atmosphere and the agricultural land, not considering energy-related emissions, is estimated to be approximately balanced at around 0.04  $GtCO_{2eq}$  yr<sup>-1</sup> (Smith et al., 2007), while energy-related CO<sub>2</sub> emissions accounted for around 9% of global agricultural greenhouse gases<sup>37</sup> in 2000 (Baumert et al., 2005), although this share can be higher in industrial countries within intensive agricultural systems.

Moreover, land-use change and forestry (LUCF) are responsible for around 13% of global anthropogenic greenhouse gas emissions, i.e. some 5.4 GtCO<sub>2eq</sub> yr<sup>-1</sup> on average during the period from 2000-2005, respectively (Houghton, 2008). The carbon flux includes emissions due to land clearing, emissions from forest products (including woodfuel) (80% in Duxbury and Mosier, 1993), and emissions from the oxidation of soil organic matter in the years following initial cultivation on former forest land (20% in Duxbury and Mosier, 1993). On the other hand, the carbon sinks accounted for are re-growing forest (vegetation and soils) after agricultural abandonment, reforestation, harvest and fire suppression. The assessed flux does not take into account the influence on carbon stocks of agricultural or silvicultural practices that do not imply changes in area, such as changes in species, no-till agriculture and thinning of forests, for instance. Finally, the assessment does not consider the indirect effects of fertilisation by N deposition or increased atmospheric concentration of CO<sub>2</sub>, that could partly counterbalance the rise in  $CO_2$  emissions (Houghton, 2003).

Agriculture is indirectly responsible for a large part of these emissions due to land clearing to convert lands into croplands or grasslands, primarily deforestation in developing countries driven by the conversion of forest to agricultural lands (Houghton, 2003; Baumert et al., 2005); 60% of released carbon due to land-use change between 1850 and 2000 came from the tropics and during the 1990s, the net carbon flux outside the tropics has actually turned into a net sink (Houghton, 2003). Nevertheless, anthropogenic CO<sub>2</sub> emissions from land-use change and the forestry sector are subject to extraordinary uncertainties<sup>38</sup>, notably linked to the varying availability and quality of regional land-use data and to uncertainties in estimating forest growth rates and carbon stocks in ecosystems affected by various human management practices (Houghton, 2003). Thus, estimates of the carbon fluxes are uncertain in the order of  $\pm 150\%$  for large fluxes, and  $\pm 50$  MtC yr<sup>-1</sup> for estimates near zero (Houghton, 2003).

For the 1990s, IPCC estimates of  $CO_2$  from land-use change ranged between 12% and 28% of world total  $CO_2$  emissions (in Baumert et al., 2005). This sector also includes N<sub>2</sub>O and CH<sub>4</sub> emissions, although no reliable global estimates make it possible to assess the share of these emissions that are linked to land-use change and forestry (Baumert et al., 2005). In order to assess the greenhouse gas emissions due to

<sup>&</sup>lt;sup>36</sup> These shares are followed in the report by the mention "*medium agreement, medium evidence*", and the same for the balanced  $CO_2$  net flux by agricultural soils "*low agreement, lim-ited evidence*".

 $<sup>^{37}</sup>$  The remaining 6% of agricultural greenhouse gases by subsector are undifferentiated sources of CH<sub>4</sub> and N<sub>2</sub>O.

<sup>&</sup>lt;sup>38</sup> As written by the authors Baumert et al., 2005, p. 91.

agricultural production better, a deeper understanding of the local determinism of  $CO_2$ ,  $N_2O$  and  $CH_4$ contributions is needed, which also takes into account the impacts of land-use change and agricultural practices on these emissions. We focus here, within the framework of biofuels from energy crops, on  $N_2O$ and  $CO_2$  emissions.

# 5.2.2 N<sub>2</sub>O Emissions

#### **Global Budget**

N losses from agricultural fields are a very critical issue for mainly two reasons. First, they represent a net loss of nutrient for the plant and a consequently supplementary cost in terms of fertilisation. Second, all the Nr<sup>39</sup> leaks outside of the soil-plant system are sources of pollution. The mechanisms of N losses are diverse, and so are their impacts on the environment. The determinism of N losses and the characterisation of their impacts are complex especially because of difficulties in considering changes in spatial and temporal scales between emission sources and final impacts, as reactive Nr is widely dispersed by hydrologic and atmospheric transport (Galloway et al., 2003). Furthermore, little is known about how to quantify synergic or antagonistic processes occurring between midpoint impact (e.g. acidification) and endpoint damage (e.g. water toxicity), inducing further uncertainty in indirect emissions that are linked to primary direct emissions.

A wide range of experiments and studies have been focusing on how to improve N fertilisation efficiency in order to firstly reduce the source of these losses as far as possible. In this sense, much progress has already been achieved during the last decades, notably by better adapting the type, doses and applications of fertilisers to the crop needs and pedo-climatic conditions. However, many questions still remain, especially concerning the determinisms of gaseous N losses at the field scale. The imbalance between total inputs and outputs of N in agricultural systems has puzzled scientists for more than 50 years (Wrage et al., 2001). <sup>15</sup>N balances show deficits in N fertiliser recovery that vary between 1% and 35% (Recous et al., 1988).

Special attention has increasingly been paid to nitrous oxide (N<sub>2</sub>O), since it is an important agricultural greenhouse gas. Indeed, due to its long residence time in the atmosphere and its high relative absorption capacity per mass unit, its 100-year global warming potential is about  $298^{40}$  times that of CO<sub>2</sub> per mass unit (Forster et al., 2007). Considering its current concentration in the atmosphere, N<sub>2</sub>O is the fourth largest single contributor to positive radiative forcing (after CO<sub>2</sub>, CH<sub>4</sub> and tropospheric ozone<sup>41</sup>) (Denman et al., 2007). Its radiative forcing averaged 8.5% of total radiative forcing for the period 1750-2000, when CO<sub>2</sub> contributed to 85% of this total radiative forcing (Forster et al., 2007). N<sub>2</sub>O is also the main source of stratospheric  $NO^{42}$ , that catalyses the photolysis of  $O_3$ (Conrad, 1990).

Nitrous oxide is naturally produced in soils through the microbial processes of denitrification and nitrification. Nitrification is the aerobic oxidation of ammonium into nitrate, and denitrification is the anaerobic reduction of nitrate into nitrogen gas (N<sub>2</sub>). N<sub>2</sub>O is an obligate intermediate in the reaction sequence of denitrification and a by-product of nitrification (IPCC, 2006). Hence, the availability of inorganic Nr in the soil appears to be one of the main controlling factors of these reactions, and the intensification of agricultural activities leading to more use of N fertilisers enhances N<sub>2</sub>O emissions by the soils (Mosier et al., 1998; IPCC, 2006).

 $<sup>^{39}</sup>$  Nr means reactive nitrogen compounds, i.e. all inorganic and organic N compounds except  $N_2,$  that is a non-reactive N compound.

 $<sup>^{40}</sup>$  The former GWP in the second IPCC assessment report was 310 eq CO<sub>2</sub> per kg, 298 includes the indirect negative radiative forcing due to the destruction of stratospheric ozone.

<sup>&</sup>lt;sup>41</sup> Radiative forcing (W.m<sup>-2</sup>), or global warming potential, refers to the change in the radiative balance on Earth's surface that is normally ensured by the natural greenhouse effect whose dominant contributing gases are water vapour (60–70% in Duxbury and Mosier, 1993), CO<sub>2</sub> (25% in Duxbury and Mosier, 1993) and O<sub>3</sub>. A positive radiative forcing (warming) occurs when the concentration of greenhouse gases increases; a negative radiative forcing (cooling) when precursors that lead to the destruction of greenhouse gases are released into the atmosphere. Halocarbons are also main contributors to radiative forcing to an extent similar to that of tropospheric ozone (Forster et al., 2007). They are not mentioned amongst the first single contributors though, because they encompass several gas contributors.

 $<sup>^{42}</sup>$  NO<sub>x</sub> = NO + NO<sub>2</sub> which are in photochemical equilibrium. NO<sub>x</sub> is mostly firstly emitted in the form of NO (Conrad, 1990). NO<sub>x</sub> is a common anthropogenic pollutant (Duxbury and Mosier, 1993).



**Fig. 12** Global budgets of  $N_2O(14)$  and  $CO_2(15)$  (Denman et al., 2007)

Agriculture is the single biggest anthropogenic N<sub>2</sub>O source (Denman et al., 2007), being the third overall most important source after soils under natural vegetation, especially land at tropical latitudes due to more rapid N cycling (Duxbury and Mosier, 1993), and N<sub>2</sub>O release by oceans. Compared with  $CO_2$  (Fig. 12), there is only one significant known sink of N<sub>2</sub>O, i.e. its destruction in the stratosphere after an average residence time of 114 years in the atmosphere, and no robust evidence of soil N<sub>2</sub>O sink strength. The amount of N<sub>2</sub>O that is absorbed by soils, i.e. "consumed" by denitrification, is subject to extreme uncertainty. Net N<sub>2</sub>O uptake by soils has been observed under different conditions, making it difficult to identify a set of conditions promoting N2O uptake. However, factors opposing diffusion of N<sub>2</sub>O in soil generally seem to increase its consumption, as well as low Nr and O<sub>2</sub> availability<sup>43</sup> (Chapuis-Lardy et al., 2007). N<sub>2</sub>O uptake is often masked by larger N<sub>2</sub>O production and may be indirectly accounted for in global budgets, provided that emission factors are based on all measured fluxes without discarding negative measurements.

More understanding of  $N_2O$  consumption by soils is needed to take into consideration its contribution to the global  $N_2O$  budget better; especially since the current global estimated sources and sinks of  $N_2O$  are not balanced (Chapuis-Lardy et al., 2007; Goldberg and Gebauer, 2008). Reported sources are larger than summed sinks and atmospheric increase. Considering that oceanic N<sub>2</sub>O source may be underestimated<sup>44</sup> by at least two-fold (Bange, 2006), it is additionally likely that some N<sub>2</sub>O source is overestimated or N<sub>2</sub>O sink underestimated (Goldberg and Gebauer, 2008). Moreover, N<sub>2</sub>O stratospheric lifetime seems to be shorter than previously thought, which also indicates that sinks may have been underestimated (Chapuis-Lardy et al., 2007). Recent bottom-up and top-down estimates of total N<sub>2</sub>O sources in the 1990s agreed on averages of 17.7 (8.5–27.7) and 17.3 (15.8–18.4) MtN yr<sup>-1</sup>, respectively (Denman et al., 2007). With a bottom-up approach, total direct and indirect annual N2O emissions from agricultural soils in the 1990s, including synthetic fertiliser, manure and biological N fixation would average 3.5 MtN yr<sup>-1</sup> (2% of N input) (Mosier et al., 1996), 4.2 MtN yr<sup>-1</sup> (Mosier et al., 1998; i.e. IPCC, 1997) or 5.4 MtN yr<sup>-1</sup> (Denman et al., 2007). A FAO statistical model, considering most of the factors<sup>45</sup> influencing median values of N<sub>2</sub>O measurements, gives estimates

 $<sup>^{43}</sup>$  Soil humidity favours denitrification up to N<sub>2</sub>O reduction, while NO<sub>3</sub><sup>-</sup> is preferred as an electron acceptor over N<sub>2</sub>O (Granli and Bøckman, 1994).

<sup>&</sup>lt;sup>44</sup> Bange emphasised that estimates used in global budgets are out of date. Moreover, due to increased release of anthropogenic Nr into the ocean, N<sub>2</sub>O emissions by marine microorganisms could increase up to 1.6 MtN-N<sub>2</sub>O per year (in Galloway et al., 2008).

<sup>&</sup>lt;sup>45</sup> Climate, crop type, fertiliser type, application rate, mode and timing of application, soil organic C and N content, soil pH, soil texture and drainage, measurement technique, frequency of measurements, length of measurement period. This analysis does not include organic soils; neither did the one from Mosier et al. (1996). Organic soils are considered in the IPCC guidelines. They appear to be a great source of N<sub>2</sub>O, because of high soil organic content and low drainage, which implies reducing conditions (IPCC, 2006). Total areas of organic soils (histosols) ~1.2% of ice-free land area (online 03.02.2009: http://soils.ag. uidaho.edu/soilorders/histosols.htm).



**Fig. 13** IPCC assessments of total annual N<sub>2</sub>O emission during the 1990s. Drawn from IPCC data in Mosier et al., 1998 and Denman et al., 2007. Note: for the 1997 assessment 0.9 Tg yr<sup>-1</sup>(Mt yr<sup>-1</sup>) of emissions by agricultural soils are subtracted to the total 6.3 Tg yr<sup>-1</sup>(Mt yr<sup>-1</sup>) to prevent dou-

ble counting with part of indirect agricultural emissions already accounted for within emissions by oceans. Emissions from human excreta, rivers and atmospheric deposition can also be partly considered as agricultural indirect emissions

for total direct N<sub>2</sub>O emissions<sup>46</sup> of 3.5 MtN yr<sup>-1</sup>, with a 34% share occurring in developed countries, respectively (i.e. 3.3–3.4% of N input) (FAO, 2001). Total top-down assessed agricultural N<sub>2</sub>O emissions of 4.3 to 5.8 MtN yr<sup>-1</sup> (3.8–5.1% of N input) (Crutzen et al., 2008) are to be compared with the bottom-up estimate 6.3 MtN yr<sup>-1</sup> (Mosier et al., 1998) that also encompasses 2.1 MtN yr<sup>-1</sup> from animal waste management. IPCC assessments of global N<sub>2</sub>O budgets have been continuously evolving, notably due to improvement in considering the diverse direct and indirect sources and refining emission factors; still, the uncertainty on anthropogenic N<sub>2</sub>O remains remarkably significant (Fig. 13).

#### **Origins of Uncertainties**

Uncertainties in estimating  $N_2O$  emissions from agricultural fields originate in the difficulties (1) of identifying all the primary sources; the contribution of biological N fixation<sup>47</sup> is especially hard to quantify (Mosier et al., 1998; IPCC, 2006), (2) of following the fate of nitrogen throughout the whole nitrogen cascade that implies several processes and "actors" (Duxbury and Mosier, 1993; Galloway et al., 2003), and (3) of capturing and characterising the spatial and temporal high variability in emissions (Parkin, 1987; Mosier et al., 1996). This variability is due to multiple involved processes that each respond differently to various environmental and soil factors (Farquharson and Baldock, 2008). Moreover, these factors can interfere at three control levels: (1) in the rate of nitrification and denitrification, (2) in the proportions between the gaseous end products of these reactions, and (3) in the consumption of these gases in the soil before escaping to the atmosphere (Firestone and Davidson, 1989).

On a global basis, about 120 MtN from new Nr (fertilisers and cultivation-induced biological fixation) and 50 MtN from previously created Nr (crop residues, deposition, etc.) are added annually to agroecosystems (Galloway et al., 2003). Within the primary cycle of Nr (dashes lines in Fig. 14), only half of the N input is harvested in the crop (Duxbury and Mosier, 1993; Galloway et al., 2003), while the other half is lost by

<sup>&</sup>lt;sup>46</sup> Data for fertilisation IFA/IFDC/FAO (1999), land-use in FAO, 2001.

 $<sup>^{47}</sup>$  Symbiotic rhizobia in root nodules are able to denitrify. This can lead to N<sub>2</sub>O emissions, possibly 4 kg N.ha<sup>-1</sup> for improved pastures; legumes could increase N<sub>2</sub>O emission two- to three-

fold compared with unfertilised fields in Mosier et al. (1996). This denitrification by rhizobia could also lead to net  $N_2O$  consumption depending on local factors.



Fig. 14 A simplified flow of fertiliser N through the environment (Duxbury and Mosier, 1993) and  $N_2O$  emissions with ranges into brackets ( $N_2O$ -Nkg) estimated with the IPCC Tier

1 method (IPCC, 2006) considering some US average data on the diverse fractions as given by Duxbury and Mosier (1993)

a combination of leaching (19–26% of input in Smil, 1999), run-off and gaseous losses through direct emissions (15–35% of input in Smil, 1999), primarily from denitrification (Duxbury and Mosier, 1993), volatilisation and nitrification. Secondary N flows, shown by the solid lines (Fig. 14), encompass N<sub>2</sub>O emissions from nitrification and denitrification through two indirect pathways; (i) following volatilisation of NH<sub>3</sub> from urea, ammonia or manure application, of NO<sub>x</sub>, and the subsequent re-deposition of these gases and their products NH<sub>4</sub><sup>+</sup> and NO<sub>3</sub><sup>-</sup> into soils and waters; and (ii) after leaching and run-off of Nr, mainly NO<sub>3</sub><sup>-</sup>.

Throughout the whole Nr life cycle, only a small amount (about 4 MtN from the initial 170 Mt in Smil, 1999) will accumulate in the agroecosystems, while the rest will eventually transfer back into the atmosphere (Duxbury and Mosier, 1993; Galloway et al., 2003), including the 21 MtN temporarily stored through human consumption of grain (64%), and meat (CAFOs<sup>48</sup> 20%) (Galloway et al., 2003). Despite current knowledge, it is still not possible to reliably

predict the fate of a unit of Nr that is applied or deposited in agroecosystems (Mosier et al., 1996), and the total amount of Nr lost through denitrification in agroecosystems is poorly known. Agroecosystems receive about 75% of the Nr created by human activity (Galloway et al., 2003). In the mid-1990s, the fate of only 35% of Nr inputs in the terrestrial biosphere was relatively well known: 18% was exported to and denitrified in coastal ecosystems, 13% deposited into the ocean, and 4% directly emitted as N<sub>2</sub>O; the remaining 65% either accumulated in soils, vegetation and groundwater or was denitrified into N<sub>2</sub>, but the uncertainty of those estimates remains large on every scale (Galloway et al., 2008), and further uncertainties appear when trying to assess all the direct and indirect N<sub>2</sub>O emissions. In the field, direct N<sub>2</sub>O emissions from N-fertilised agricultural fields have been found to vary between 0.001% and 6.8% of the N applied. Case studies combining diverse measurement techniques confirmed that uncertainty in N2O fluxes found in the literature was indeed due to diverse combinations of controlling factors and not linked to the analytical methods (Mosier et al., 1996).

<sup>&</sup>lt;sup>48</sup> Concentrated animal feeding operations.

#### **Emission Factors**

The 2006 IPCC guidelines to assess N<sub>2</sub>O emissions from managed soils (IPCC, 2006) consider all identified direct N<sub>2</sub>O sources, except direct emission from biological fixation due to the lack of experimental evidence, and the two above-mentioned indirect pathways. The Tier 1, 2 and 3 methods consist first of a comprehensive accounting of all N input into the fields, including inorganic and organic fertilisers, as well as mineralised N from soil organic carbon due to landuse change or crop residues that indicate background emission levels linked to recovery from past managements. These Nr amounts are then multiplied by default emission factors (Tier 1), emission factors related to country-specific data when available (Tier 2), or the emissions are estimated with process-based models (Tier 3). Following the same order, Tier 1, 2 and 3 methods guide the estimation of input fractions to be multiplied by indirect emission factors with less to more country specificity, respectively.

Following the Tier 1 method and the given default emission factors, the 100 kg of Nr input on a US field (assuming maize for crop residues) as characterised by Duxbury and Mosier (1993) (Fig. 14) would emit 1.86 kg N-N<sub>2</sub>O with a wide uncertainty range between 0.47 kg and 12.4 kg. These estimates include emissions from crop residues, without time lag, and emissions due to a secondary cycle through manure recycling and further run-off and leaching due to its application. Using the 1.25%  $\pm$  1% of N input lost as N<sub>2</sub>O (Bouwman, 1994) and further 0.75% of this input lost through indirect emissions, the same scenario would be expected to lead to 2 kg N-N<sub>2</sub>O  $\pm$  1 kg emissions (Mosier et al., 1996). The rough 2.5 kg N-N<sub>2</sub>O estimated by Duxbury and Mosier (1993) did not include all indirect contributions.

Statistical models aim at finding reproducible correlations, i.e. relationships representative of most data sets, between controlling factors and emissions, e.g. emission factors depending on the type and amount of applied fertiliser. In this sense, the more data collected on the different direct and indirect emissions, the more informative emission factors will be. The multiplication of data sets and the development of complementary techniques on various scales (aircraft measurements, micrometeorological techniques, chambers (see picture below), <sup>15</sup>N balance, C<sub>2</sub>H<sub>2</sub> inhibition and lab work), have already made it possible to improve the understanding of controlling factors beyond a coarse linearity with N fertiliser. It has also been proven that measurements should be carried on throughout the year, because maximum fluxes were observed at different times for different treatments, and with a high frequency to capture temporal emission patterns (Mosier et al., 1996; Laville et al., 1997; Beheydt et al., 2007; Pattey et al., 2007). Frequent measurements appeared to yield lower total emissions (FAO, 2001; Pattey et al., 2007) due probably to less error with interpolation of punctual high emissions (Conen et al., 2000). Still, increased numbers of measurements that represent a wider range of agricultural systems are needed in order to improve statistical models (Mosier et al., 1996; Stehfest and Bouwman, 2006), as well as processbased ones.



Measurement of N<sub>2</sub>O and CO<sub>2</sub> fluxes with automatic chambers in a sugar beet plot, Estrees-Mons, June 2008 bessou©INRA

Given the complexity of N2O emission control (Fig. 15), it is almost impossible to determine a quantitative relationship between the cause (a change in any ecological driver or environmental factors) and the consequence (N<sub>2</sub>O fluxes) through simple correlation or regression analysis (Li, 2007). The extreme spatial and temporal heterogeneity of many primary drivers actually obscures the relationships between cause and effects for many of the biogeochemical processes, so that correlations between a change in primary drivers and linked changes in biogeochemical cycles are inherently non-linear (Li, 2007; Conrad, 1996). As regression models neglect several variables, because datasets used for developing the model did not distinguish these variables, for instance, emission factors cannot in essence lead to significant reduction



**Fig. 15** Scheme of the determinism of N<sub>2</sub>O emissions by soils adapted from Li (2007); Wrage et al. (2005); Farquharson and Baldock (2008). Soil particles, gaseous and aqueous phases ( $\phi$ ) are artificially "well distinguished". The two extremities of the symbolised pore represent the pore continuum throughout the soil matrix. Reactions take place at the interface of soil particles

of estimation uncertainty and cannot always be used to test different management or mitigation scenarios (Beheydt et al., 2007).

## Process-based Models

Process-based models make it possible to assess emissions with more accuracy, because of a better accounting for all involved processes and local conditions. Numerous models can nowadays simulate cropping systems and the associated fluxes between the soilplant-atmosphere compartments. By simulating plant uptake, biomass growth and residues, nitrate leaching or volatilisation, for instance, models can provide insight into the amounts of Nr that might be directly or indirectly emitted as N<sub>2</sub>O. Specific sub-models then simulate the part of these amounts that is expected to

and aqueous phase where microorganisms and substrates are. Denitrification takes place in the "anaerobic dashed-line rectangle". Dashed arrows lead to  $N_2O$  emissions by nitrifier denitrification occurring in low  $O_2$  conditions, which implies that some  $NH_3$  oxiders process the whole reaction chain (Wrage et al., 2001)

be emitted as  $N_2O$ . Dynamic models for  $N_2O$  emission in relation to soil processes have been available for a dozen years and in more recent years developed for different ecosystems and N species (Sutton et al., 2007).

N cycling models can be classified following three approaches: (1) simplified empirical process models in which N cycling processes are assumed to be determined by easily measurable parameters; (2) microbial growth models, where N dynamics is simulated by explicitly representing the dynamics of involved microorganisms; and (3) soil structural models simulating physical processes such as diffusion into and out of soil aggregates where occurring anoxia leads to denitrification (Parton et al., 1996). Heinen (2006a) compared some 50 process-based denitrification models. Most simplified models are comparably based on a potential denitrification ( $D_p$ ) weighted by a product of reduction functions due to nitrate content, degree of saturation, soil temperature and soil pH. The potential denitrification represents the soil microorganisms' capacity to reduce nitrates under non-limiting conditions, i.e. depending on the soil organic carbon content and microorganism populations. It can be measured by reproducing these optimal conditions on intact soil cores, for instance (Hénault and Germon, 2000), or deduced from  $CO_2$  measurements that show the microorganisms' activity.

There is no consensus on the diverse reduction functions that are empirical and were calibrated from site-specific studies. Hence a universal simplified denitrification model is unlikely to exist and a chosen simplified model can only be used provided that parameters are calibrated for each location, with particular attention paid to determining the parameters of the saturation function, to which the model is the most sensitive (Heinen, 2006a).

$$D_{a} = D_{p} f_{N} f_{S} f_{T}$$

$$= D_{p} \underbrace{\frac{N}{K+N}}_{f_{N}} \underbrace{\left(\frac{S-S_{t}}{S_{m}-S_{t}}\right)^{w}}_{f_{T}} \underbrace{\mathcal{Q}_{10}^{(T-T_{r})/10}}_{f_{T}} \quad (2)$$

 $D_{\rm a}$  is the actual denitrification rate (mg N kg<sup>-1</sup> d<sup>-1</sup> or kg N ha<sup>-1</sup> d<sup>-1</sup>),

 $D_{\rm p}$  is the potential denitrification rate (mg N kg<sup>-1</sup> d<sup>-1</sup> or kg N ha<sup>-1</sup> d<sup>-1</sup>),

 $f_{\rm N}$  is a dimensionless reduction function for N, N is the nitrate content (mg N kg<sup>-1</sup> or L<sup>-1</sup>), K is the nitrate content where  $f_{\rm N} = 0.5$ ,  $f_{\rm S}$  is the dimensionless reduction function for the dimensionless degree of saturation S,  $S_{\rm m}$  is S above which  $f_{\rm S} = 1$  (in the remainder of this paper  $S_{\rm m} = 1$ ), St is a threshold value for Sbelow which no denitrification occurs ( $f_{\rm S} = D_{\rm a} = 0$ ),  $f_{\rm T}$  is a dimensionless reduction function for soil temperature T,  $T_{\rm r}$  is a reference T where  $D_{\rm p}$  is determined at (mostly)  $T_{\rm r} = 20^{\circ}$ C, and  $Q_{10}$  is an increase factor for a 10°C increase in T. (Heinen, 2006b).

Testing a common simplified model (Eq. (2)) with eight Dutch data sets, the latter author showed that parameters differed across location and that no aggregation could be done based on soil type. The optimisation could not result in perfect prediction on the point scale, and was only good for cumulative denitrification for sand and loam soils, as under- and overestimations seemed to counteract in the long term (Heinen, 2006b). The model is very sensitive to errors in the estimates of the parameters. These errors (computed for 250 soil conditions and 25 parameter conditions) propagate in the prediction of denitrification, so that defining parameters with 10% accuracy would lead to a coefficient of variation in the relative denitrification rate of about 10% (Heinen, 2006a). A test on error propagation on parameter estimation with an artificial data set showed that w and  $Q_{10}$  were overestimated, while  $S_t$  and K were underestimated, with large coefficients of variation; the greatest uncertainty was found for K (Heinen, 2006b).

The NOE model (Hénault et al., 2005) uses similar equations to equation (2) to simulate denitrification and nitrification with Michaelis-Menten functions of nitrate and ammonium, respectively, two different saturation functions, and a common temperature function. The nitrification potential is not introduced explicitly. The NGAS model (Parton et al., 1996, 2001) is also an empirical denitrification and nitrification model that was developed using laboratory- and field-observed gas fluxes from different soils. N2O fluxes are simulated using simple relationships controlled by soil saturation, texture, temperature, pH, respiration, and NO<sub>3</sub><sup>-</sup> and  $NH_4^+$  contents. Comparing these two sub-models, both coupled within the CERES crop model, NGAS appeared to be easier to operate as no site-specific data are needed; however, it was therefore also less accurate than NOE (Gabrielle et al., 2006).

Another comparison of four models simulating N cycling, CENTURY-NGAS (Del Grosso et al., 2001), DNDC (Li et al., 1992; Li, 2000), Nexpert (Engel and Priesack, 1993) and NASA-CASA (Potter et al., 1997), also showed discrepancies amongst results. Although all four models generally agreed on global N cycling rates, they presented a wide range of results concerning the different gas fluxes for both cumulated totals and temporal patterns; even when models agreed on N<sub>2</sub>O emissions, then N<sub>2</sub>, NO<sub>x</sub> or NH<sub>3</sub> fluxes diverged (Frolking et al., 1998). An accurate modelling of soil moisture dynamics and the response of modelled denitrification to soil moisture appeared to be a key for reliable N<sub>2</sub>O emissions (Frolking et al., 1998). Site-specific NOE parameters thus explained the performances better than NGAS. Nevertheless, site-specific parameters are not easily available, and pedotransfer functions would be needed to infer these parameters from basic soil characteristics (Gabrielle et al., 2006).

Up-scaling empirical models may come at the expense of prediction accuracy, whereas mechanistic models rely more on deterministic relationships based on fundamental knowledge on the interaction of predictor variables and modelled processes, which does not depend on the site or the study scale. Although mechanistic model structure is defined from the process knowledge, numerical fitting is also often used to parameterise such models, adding elements of empiricism (Farquharson and Baldock, 2008). DNDC is a microbial growth model that has been widely used to simulate N<sub>2</sub>O emissions from diverse ecosystems; it has even been upgraded to deal with peak N2O production due to freezing and thawing events (Pattey et al., 2007). This model tracks microbial activities in soils by computing the Nerst and Michaelis-Menten equations that describe interactions between microbial activities and the driving factors: soil redox potential, soil organic carbon and electron acceptors. At each time step, the concentration of oxygen and other electron acceptors determines the soil Eh and the consequent anaerobic volumetric fraction into which substrates are proportionally allocated. This defines conditions for microbial activities and the following substrate consumption that ends up with a change in soil Eh looping to the next time step (Li, 2007). Extensive testing of DNDC versus field measurements has demonstrated its ability to simulate N<sub>2</sub>O and NO emissions but still with some significant uncertainty, which is at least partly due to the still limited knowledge about the ecosystem processes involved in C and N cycling (Sutton et al., 2007). This is confirmed by a test of DNDC with 22 long-term N<sub>2</sub>O measurements. When N<sub>2</sub>O simulations were in agreement with field measurements, the patterns of  $NH_4^+$  and/or  $NO_3^-$  were not captured by the model or vice versa. In general, DNDC gave higher and more frequent N<sub>2</sub>O peaks, leading to an average overestimation taking all measurements into consideration, and both (large) under- and overestimations when looking at individual results. Although statistics indicated that simulations were not optimal, the general agreement between simulated and measured N<sub>2</sub>O total losses was better than with the three tested regression models, including Bouwman's emission factor  $(1.25\% \pm 1)$  used in the IPCC Tier 1 (1996). Improvement of the model would be necessary to use easily available data such as  $NO_3^-$  as a response variable and test mitigation scenarios, without having to measure N<sub>2</sub>O to validate the model (Beheydt et al.,

2007) Moreover, there are still recognised factors for modelling  $N_2O$  emissions that are not adequately understood (Farquharson and Baldock, 2008).

More precision on N<sub>2</sub>O emissions is essential for compliance with the precise mandatory reduction targets. How to consistently reach a 10 or 20% greenhouse gas reduction target when emission estimate uncertainty is larger than 10 or 20%? Within the framework of local LCAs, the simulation of N<sub>2</sub>O emissions with a process-based model may drastically reduce the uncertainty compared with emission factors, provided that the model performs correctly on the local scale, i.e. that parameters are well calibrated for the cultivation sites. Local LCAs might be the uniquely consistent scale to produce useful estimates of N<sub>2</sub>O emissions, as long as the understanding of all controlling factors remains insufficient to improve process-based models, which would make it possible to produce robust estimates on various scales.

#### CO<sub>2</sub> Emissions and Land-use Change in Local LCA

Carbon may accumulate in soils, mainly in organic form. The removal of atmospheric C by plants and storage of fixed C in stable fractions of soil organic C (SOC) is termed "soil C sequestration" (Lal, 2004a). Hence, SOC comes from dead plant parts (leaves, roots, etc.), plant rhizodeposition and organic matter applications (animal waste, etc.). However, its storage is not definitive because dead organic matter undergoes a series of biogeochemical transformations, including decomposition, and is eventually mineralised by microorganisms and released as CO2 or CH4, through respiration or fermentation, respectively (Arrouays et al., 2002). Carbon stock in soil is the result of a dynamic balance between "inputs" of organic matter and "outputs" due to mineralisation, erosion, leaching or combustion. The soil may act either as a carbon source or as a carbon sink, according to the ratio between inputs and outputs. Carbon inputs depend on primary production (controlled by edaphic factors such as solar radiation, temperature, water and nutrient availability), and organic matter returned to the soils (e.g. crop residue management). Carbon outputs depend on biotransformation rates, controlled by the organic matter composition and local physico-chemical conditions (temperature, moisture, oxygen, etc.). Biotransformation is also slowed down when organic matter is associated with mineral particles (particularly clay) which provide "physical protection" against the activity of microorganisms (Balesdent et al., 2000).

## Impact of Land-use Change on Soil Organic Carbon

On the global scale, the SOC pool is about 1550 GtC (+950 GtC inorganic carbon), exceeding by far the atmospheric (760 GtC) and biotic pools (560 GtC) (Lal, 2004a,b). Comparatively, geologic and oceanic pools represent 5000 GtC and 38 000 GtC, respectively (Lal, 2004a). Over the last 200 years or so, soils may have lost between 55 and 78 GtC because of land-use conversion and soil cultivation. On a global scale, the SOC pool to 1-metre depth has a predominant range of 50 to 150 tC ha<sup>-1</sup>, but can reach 800 tC ha<sup>-1</sup> in organic soils (Lal, 2004b). The SOC pool varies widely among ecological regions, being higher in cool and moist than warm and dry regions, and among ecosystems, SOC pools to 1-metre depth average for croplands 80–103 tC ha<sup>-1</sup>, temperate grasslands 141–236 tC ha<sup>-1</sup>, tropical, temperate and boreal forests, 122, 96-147, and 247-344 tC ha<sup>-1</sup>, respectively (Lal, 2004a). SOC stocks are higher in cool temperate forests and wetlands where plant productivity is relatively high but the activity of soil microorganisms is slowed down by the temperature, whereas SOC stocks are lower in the wet tropics where organic matter turnover is rapid, and lowest in dry regions where plant growth is limited (Cowie et al., 2006).

Arrouays et al. (2001) provided estimates of average SOC stocks in France (to a depth of 30 cm), according to land use. Arable lands are characterised by relatively low stocks: 43 tC ha<sup>-1</sup> on average, whereas permanent grasslands and forests (excluding litter) exhibit average stocks of nearly 70 tC ha<sup>-1</sup>. These differences can be explained partly by a greater supply of carbon to the soil under grassland and forest (mainly from the roots but also from shoot litter), and partly by a shorter residence time of carbon under arable land (Soussana et al., 2004). Increased biodegradation rates in arable land may be due to multiple factors, e.g. changes in soil climate, nutrient availability and pH. A major factor would be the double impact of soil tillage that directly enhances mineralisation through increased oxygenation and de-protection of the organic matter by soil tillage (Balesdent et al., 2000; Germon et al., 2007). Indeed, a fraction of the organic matter included

in micro-aggregates is physically protected from biodegradation, and inversely contributes to the aggregate cohesion through the binding of mineral particles by organic polymers and to the water stability of aggregates due to increased hydrophobicity (Chenu et al., 2000). Disruption of soil structure by tools and subsequent disruption of micro-aggregates by the action of rain expose the hitherto encapsulated C to biodegradation (Balesdent et al., 2000; Lal, 2004a). As organic matter is removed and dissolved from top soils, aggregates also become less stable, which could lead to synergetic losses of organic matter (Germon et al., 2007). A great part of carbon supply under grasslands is also due to larger root turnover and rhizodeposition than under arable crops. This process favours carbon storage because direct incorporation into the soil matrix leads to a higher stabilisation by physical protection and root litter is also chemically more stable (Soussana et al., 2004). In particular, grasses whose roots reach the deep part of the soil profile well below the plough layer make it possible to sequester C that is less prone to oxidation and loss (Fisher et al., 1994).

Kinetics of SOC accumulation or release following land-use change is non-linear and asymmetrical (Arrouays et al., 2002; Seguin et al., 2007). Variations are more rapid during the first years after land-use or land-management change and reach a plateau after several decades. The time taken to reach this new equilibrium (sink saturation) is highly variable, around 100 years in a temperate location, up to several centuries in boreal regions. Moreover, if the land-use change is reversed, the accumulated SOC will be lost, usually more rapidly than it was accumulated (Smith, 2004). This could be partly explained by the synergetic effect mentioned above. Arrouays et al. (2002) provided an estimation of mean carbon changes due to land-use change in France (Fig. 16). This estimate is based on the use of an exponential function, fitted with data available in the literature and French average soil C stocks at equilibrium for the main types of land use. According to this study, the mean carbon change implied each year over a 20-year period by converting forest to annual crop is about -0.75 tC ha<sup>-1</sup>, and  $-0.95 \pm 0.3$  tC ha<sup>-1</sup> by converting permanent pasture to annual crop. On the contrary, conversion of arable land to forest or grassland leads to a mean annual soil carbon storage of 0.45  $\pm$  0.25 tC ha<sup>-1</sup> for forest and 0.5  $\pm$  0.25 tC ha<sup>-1</sup> for grassland, over a 20-year period.



Fig. 16 Change in soil carbon stocks associated with land use change. These are modal values for mainland France. The 95% confidence interval of these values is about  $\pm$  40% (Arrouays et al., 2002)

The variability of estimates is due mainly to the diversity in climatic conditions and soil characteristics (Seguin et al., 2007; Soussana et al., 2004). In a meta-analysis of 74 international publications, Guo and Gifford (2002) confirmed the high impact on SOC of grassland or forest conversion to cropland. According to this study, soil carbon stocks decrease on average by 42% after conversion of forest to crop and 59% after conversion of grassland to crop.

Introduction of perennial energy crops in current annual crop systems may increase carbon sequestration, due to the lack of soil tillage during the crops' growing cycle (typically 15-20 years), their high biomass production and pre-harvest losses, and their extensive root system (Lemus and Lal, 2005). In a field experiment in southern Quebec, Zan et al. (2001) measured a total root carbon content 4 to 5 times greater for 3-year-old SRC willow and switchgrass than for corn. High below-ground biomass (rhizomes and roots) was also measured for miscanthus. Below-ground biomass ranged, for example, from 15 to 25 tDM ha<sup>-1</sup> for 4-9-year-old miscanthus in Germany, corresponding to 7.6-10.2 tC ha<sup>-1</sup> (Kahle et al., 2001). It is thus expected that, as observed under grassland, root turnover and rhizodeposition should be a major carbon input under perennial energy crops. Perennial energy crops are usually harvested in late winter or early spring with high dry matter content. This practice causes pre-harvest losses, mainly by leaf senescence. Mean pre-harvest losses during 3 years



**Fig. 17** Observed changes in soil carbon stocks associated with the introduction of perennial energy crops (SRC poplar or willow: circles, miscanthus: triangles and switchgrass: squares/diamonds) after annual crops or grassland. Drawn from data in Garten and Wullschleger, 1999; Jug et al., 1999; Ma et al., 2000a,b; Kahle et al., 2001; Zan et al., 2001; Frank et al., 2004; Hansen et al., 2004; Clifton-Brown et al., 2007

of miscanthus cultivation were 4.5 tDM ha<sup>-1</sup> yr<sup>-1</sup>, corresponding to about 2 tC ha<sup>-1</sup> yr<sup>-1</sup>, in the same field experiment in Germany (Kahle et al., 2001). Several authors have evaluated impacts of perennial energy crops (short rotation coppice (SRC), miscanthus, switchgrass) on SOC, using field measurements in long-term experiments (Fig. 17).

The results of these experiments are highly heterogeneous, which is probably due partly to the diversity of climatic, pedological and agricultural conditions, and partly to differences in measurement methodology (e.g. soil sampling depth). However, in general terms, conversion of annual crops to perennial energy crops seems to increase carbon sequestration, which may not be the case when perennial energy crops are introduced after grassland. Also, there is no clear difference between perennial energy crops (short rotation coppice, miscanthus, switchgrass).

Land clearing (i.e. conversion of forest or grassland to arable crops) can lead to a large release of  $CO_2$ . To evaluate this effect, it is necessary to take into account the different carbon pools, i.e. not only the soil carbon pool but also the above-ground and below-ground biomass carbon pools. According to Fargione et al. (2008), the amount of  $CO_2$  released during the 50 years following land conversion would be 737 t $CO_2$  ha<sup>-1</sup> in the case of a tropical forest converted to soybean in Brazil, and 134 t $CO_2$  ha<sup>-1</sup> in the case of natural grassland converted to corn in the US. It represents a "carbon debt" that should be included in LCA when biofuels are introduced after land clearing. The additional carbon sequestration by perennial energy crops compared with annual crops should also be taken into account in LCA. However, it is important to keep in mind that any carbon sequestration in soil is finite and reversible (Powlson et al., 2005).

# Impact of Agricultural Practices on Soil Organic Carbon

Agricultural practices can modify SOC levels in arable lands, by changing the amount of carbon supply to the soil or by changing the residence time of carbon in the soil. Effects of soil management practices on carbon sequestration have been widely studied. No-tillage generally implies an increase in soil carbon levels, compared with tillage (see review by Arrouays et al., 2002; Germon et al., 2007). The mean increase in carbon sequestration due to no-tillage or reduced tillage was estimated at  $0.2 \pm 0.13$  tC ha<sup>-1</sup> yr<sup>-1</sup>, over a 20-year period. However, there is no consensus about the magnitude of the differences between conventional tillage and no-tillage: some authors reported negligible differences; others found considerable differences (Arrouays et al., 2002). Origins of this variability are not well known.

According to Balesdent et al. (2000) comparing carbon stocks between tillage and no-tillage treatments causes some difficulties, due to changes in bulk densities and carbon repartition along the soil profile. Furthermore, no-tillage may refer to different degrees of "conservation tillage", such as direct sowing or noninverting ploughing. Conservation tillage also encompasses intermediary techniques referred to as "reduced tillage", etc. Due to the complexity of soil organic matter (SOM) dynamics (Fig. 18), the varying depths and degrees of soil disturbance and the varying duration of the treatments across studies can also imply discrepancies amongst conclusions. The impact of conservation tillage on  $CO_2$  emissions also varies amongst studies.



Fig. 18 Soil organic matter is the key indicator in sustainable soil management (in Van Camp et al., 2004)

Although  $CO_2$  measurements directly after soil tillage mostly showed higher emissions than under direct sowing, impacts in the long term are less clear.  $CO_2$ long-term measurements under conservation tillage are still lacking and little is known about  $CO_2$  emissions when the new SOC equilibrium is reached under conservation tillage (Germon et al., 2007).

The use of catch crops over intercropping periods can represent an interesting option in terms of carbon sequestration. Arrouays et al. (2002) modelled a potential increase of  $0.15 \pm 0.08$  tC ha<sup>-1</sup> yr<sup>-1</sup> over a 20-year period, for an annual incorporation of catch crops. In a field experiment with spring barley in Askov (Denmark), the mean difference in SOC between no catch crop and catch crop treatments was 1 tC ha<sup>-1</sup> after 10 years<sup>49</sup>, corresponding to an annual increase of 0.1 tC ha<sup>-1</sup> yr<sup>-1</sup> due to annual catch crop incorporation in soil (Thomsen and Christensen, 2004). Crop residue management can also impact soil organic carbon. Saffih-Hdadi and Mary (2008) compiled nine well-documented long-term field experiments, which compare effects of systematic removal or incorporation of cereal straws on SOC evolution. They differed in climate, soil type, carbon input and duration (from 12 to 35 years). The measured SOC increase due to straw return (as compared with straw removal) varied from 0.078 tC ha<sup>-1</sup> yr<sup>-1</sup> to 0.385 tC ha<sup>-1</sup> yr<sup>-1</sup>, corresponding to 4.2-19.1% of added straw carbon. Climate influenced the efficiency of straw incorporation in SOC. This incorporation is much more efficient under cold climates, where it can reach up to 0.90% of the initial SOC content compared with 0.53% under warm climates. Systematic removal of straw for bioenergy purposes will then lead to a decrease in SOC content. Using a simple carbon dynamics model called AMG, Saffih-Hdadi and Mary (2008) simulated the impact of straw removal one year out of two in nine experimental sites. After 50 years, it would reduce carbon stocks by 2.5–10.9% of the initial SOC, depending principally on the experiment (soil, climate, productivity).

As SOC sequestration is provisional, it can only play a minor role in climate change mitigation. The maximum global SOC sequestration potential of  $0.9 \pm 0.3$  GtC yr<sup>-1</sup> over 50 years (Lal, 2004a) could contribute to a maximum of 2–5% towards reducing the carbon emission gap under the highest emission scenarios (Smith, 2004). However, given the drastic  $CO_2$  reduction needs to meet targets, it is already crucial that agricultural practices should aim to prevent carbon losses as much as possible, notably net  $CO_2$  emissions due to land-use change, then to implement practices that enhance SOC sequestration. A better understanding of the SOC stabilisation in deeper soil layers could also open up new options in order to increase C sequestration.

#### Land-use Impacts in LCA

"Land-use impacts are the 'amount' of land quality not present in a certain area due to the studied system, compared to a situation where the studied system had not been established" (Milà I Canals et al., 2007).

The major environmental importance of land-use impacts contrasts with the lack of consensus on this area within the field of LCA. As a result, the issue is seldom included in LCA and the credibility of LCA results is insufficient for many stakeholders. Lack of consensus comes at least partly from the failure to recognise the value judgements behind the assessment methodology. These value judgements include the following: what are the functions of land that need protection, which are the thresholds? What are the time perspective and reversible impacts? What are the future or alternative land uses? Which indicators represent the impact pathways? (Milà I Canals et al., 2007). Focusing on bioenergy chain LCAs, many studies have lately put emphasis on the necessity to develop a methodology within the LCA tool to take into account the impacts of land-use change on the ecological functions of land. Among others, the impact of deforestation on biodiversity and CO<sub>2</sub> emissions, the impact of straw removal on soil fertility, the impact of perennial crops for future land uses, etc., are examples that show the importance of such factors as part of sustainability criteria.

Nevertheless, potential impacts of land-use change are difficult to assess for mainly two reasons. First, impacts due to land-use change have to be characterised in comparison with unchanged land use. When considering agricultural land use, in principle only degradation caused by the management practice during the cultivation period should be allocated to the crop harvested (Mattsson et al., 2000). This implies defining a reference scenario and a time frame for

<sup>&</sup>lt;sup>49</sup> Mean value of the 4 straw restitution treatments (Thomsen and Christensen, 2004).

the occupation of land or recovery period. But one reference scenario is sometimes not sufficient to cover the range of possibilities when it comes to dealing with the use of new land areas or longer time frames such as in the cases of crop rotations or the cultivation of perennial crops. Second, LCA methodology based on equivalency factors is hardly adaptable for landuse change. Indeed, aggregation of parameters such as soil organic matter and landscape values, for example, is difficult so that the land-use impact category should be less aggregated than other impact categories in LCA (Mattsson et al., 2000), leading to the complexity of dealing with two approaches and a resulting mix of quantitative and qualitative information. Other approaches and tools, such as Environmental Impact Assessment, may provide more detailed information than LCA on effects of different land managements. "However, LCA is the appropriate tool to bring a life cycle perspective to support complex decisions involving different land uses, and, consequently, it should incorporate a measure of the different impact pathways affected by land use" (Milà I Canals et al., 2007).

Many references focus on suggesting indicators to include the effects of land use on productivity and biodiversity, although the practical implementation of such sets of indicators is seldom checked with a consistent framework (Milà I Canals et al., 2007). Most proposed methodologies use a number of indicators that are largely submitted by the availability of data (Anton et al., 2007). Mattsson et al. proposed to divide "land-use change" into three sub-categories: (1) soil fertility, with a set of 7 indicators, (2) biodiversity, and (3) landscape values Mattsson et al. (2000). Schenck and Vickerman (2001) also gave a list of indicators for the assessment of impacts on biodiversity. Impacts on biodiversity, for instance, are currently considered in LCA through damage to the biotic environment, or concurrence of species (Jolliet et al., 2004); the effects considered have been traditionally limited to those caused by changes in the chemical composition of the environment (toxicity, eutrophication, etc.) (Milà I Canals et al., 2007). The World Resource Institute<sup>50</sup> showed that the greatest biodiversity losses are derived from changes in land use, rather than to any chemical impacts (Schenck et al., 2001). Some of the latest methods for LCA thoroughly address land-use impacts, but fail to include effects of occupation or transformation on the resource aspect of land (Milà I Canals et al., 2007). Despite the availability of indicators, there is still a lack of consensus on which is the most ideal indicator for evaluation (Anton et al., 2007). Comparing indicators from Köllner (2001) and Weidema and Lindeijer (2001), it appears that further research is still needed to refine them so that they could deal with more specific ecosystems and geographical areas (in Anton et al., 2007).

CO<sub>2</sub> emissions due to biofuel combustion are commonly not included within the system boundary since the fuel is considered as carbon-neutral; indeed, the released carbon during combustion  $(CO_2)$  had been fixed from the atmosphere in the first place. This exclusion is hence justified when comparing a biofuel chain starting from the biomass production with a fossil fuel chain, because the actual carbon cycle is fully considered. However, when the delay between carbon capture and sequestration by the plants and re-emission is longer (land-use change, waste treatment, woody biomass, etc.) this assumption may lead to wrong conclusions. In this sense, Rabl et al. (2007) recommended that emission and removal of CO<sub>2</sub> ought to be counted explicitly at each stage of the life cycle. In this way, the LCA is furthermore consistent with the "polluter pays" principle, which implies that each greenhouse gas contribution should be allocated to the causing agent. For example, CO<sub>2</sub> from woodfuel for heating should be taxed as CO<sub>2</sub> from oil heating is, and a credit for CO<sub>2</sub> removal then only paid when and where the wood is replaced by new growth (Rabl et al., 2007). Furthermore, this stage-CO<sub>2</sub> accounting would be useful in the framework of an implementation of sustainability criteria for biofuel chains at an international level, including the accounting of a carbon debt or credit in the case of a change in SOC due to land-use or landmanagement change.

Beyond uncertainties linked to estimates of carbon stock and its changes due to land-use or management changes, the main difficulties arise from the definition of prospective land-use scenarios for comparative LCA. Areas impacted by land-use or management changes can be part of a crop rotation or more complex combined land-use patterns including indirect impacts through crop displacements. Soya in Brazil, for instance, is established on grasslands, pushing cattle

<sup>&</sup>lt;sup>50</sup> WRI: Guide to World Resources 2000–2001: People and Ecosystems: The Fraying Web of Life, Elsevier, New York, 2002.
pastures further into forests<sup>51</sup>. Therefore, dependence between impacted areas must be considered within a matrix of land-use changes across a sphere of influence that must be defined. This is particularly relevant in the case of SOC storage in order to take into account the fact that dynamics are long-lasting and often reflect transition states from past changes, and that kinetics are reversible and asymmetric. Methods to estimate the impacts must be adapted to these temporal scales, otherwise results could be biased by the approximation of impact differences between two instantaneous pictures of land-use patterns (Arrouays et al., 2002). A better accounting of the impact of land-use change on the soil quality and direct CO<sub>2</sub> emissions is necessary for biofuel LCA. Emphasis should be put on defining a harmonised methodology to include some indicators on soil quality. However, accurate estimates may be limited to local LCA given the complexity to encompass the temporal and spatial dimensions of the impact of land-use changes and the data needs. In order to simulate soil C change in bioenergy projects, it would be recommended to establish the baseline labile and recalcitrant C stocks through measurements and to model C dynamics over the land-use duration (Cowie et al., 2006). Furthermore, more research would be necessary to also introduce within LCA quantified impacts of land-use change on the albedo, surface energy balance and water cycle, and their consequences on climate change. Indeed, agriculture significantly affects climate through greenhouse gas emission and absorption, and modifications of surface properties, which act directly on different spatial scales. To date, however, the complete evaluation of the net impact of agriculture on climate through the modification of the natural environment is still not feasible (Seguin et al., 2007).

## 5.3 Biofuel Greenhouse Gas Balances

Most studies have found that the use of 1st generation biofuels results in emission reductions of 20 to 60% of  $CO_{2eq}$  relative to fossil fuels. Expected reductions for future commercialised 2nd generation biofuels are in the range of 70 to 90% of  $CO_{2eq}$  relative to fossil fuels (FAO, 2008a). The large range of emission reductions for the 1st generation biofuels is due to various types of feedstock and conversion processes, and to the different sites of production and consumption. Varying LCA assumptions also explain that greenhouse gas balances of a given biofuel chain in one region may be variable (see part 5.1). Finally, field emissions in particular are complex to assess and imply further disparities amongst studies (see part 5.2). Greenhouse gas savings are therefore often presented as ranges; it does not make much sense to give a list of mean values for each biofuel chain. However, Table 4 presents some of the main published studies to put in contrast varying results due to changes in coproduct handling, within a study and amongst studies. This overview completes the data in Fig. 11 and Table 2.

A sensitive analysis of the N<sub>2</sub>O emission factors showed that these assumptions critically impact the balance. Greenhouse gas emissions rise from 40 to 50% for methyl esters and pure vegetable oils by using Bouwman's IPCC (1995 Guidelines) emission factors instead of those from Skiba et al. (1996). The resulting greenhouse gas savings fall, for instance, to -55.5%and -66% for the rapeseed pure oil and methyl ester, respectively. Although Bouwman's factors may be more accurate because the regression was based on more data sets on a wider range of soil diversity, whereas Skiba's factors were established for the UK's soils, extrapolation from any linear model implies a high uncertainty on the results due to the site- and timedependence of field emissions. N<sub>2</sub>O emissions in the JRC/EUCAR/CONCAWE study are likely to be more accurate as they were simulated with the DNDC model (version 82N) combined with the LUCAS land-cover survey model. The resulting emission factors, moreover, include N<sub>2</sub>O indirect emissions from leached N. However, as the study assessed biofuel chains at the European level, the simulations were used to determine new emission factors through regression models. The averaged crop emission factors finally hardly give an approximate of total N2O emissions at a national level, while emissions are too variable on such a scale to help distinguish between biofuel chains and co-product options at the local level. Nevertheless, biofuel chains with a valued co-product make it possible to save a lot of greenhouse gases, especially if biomass production is optimised to reduce field emissions as much as possible.

<sup>&</sup>lt;sup>51</sup> Dr. Emily Boyd, 25/11/2005, http://www.scidev.net/en/ opinions/emissions-trading-cannot-solve-amazon-deforestatio. html.

any harmon	isation in background as	ssumptions, except fo	or units, e.g. emissions	for the fossil fuels per MJ vary acros	ss studies		
			Co-product handling				
			Mass		Energy	GHG/MJ	
			allocation	System	allocation	compared	
Biofuels	Regions	Feedstock	(% applied to biofinels)	expansion (substitutes)	(% applied to hiofitels)	with fossil finets	References
Ethanol	Brazil + shipped		(010010000	-	(	-66%	JRC/EUCAR/
	to EU	Sugar cane		Excess bagasse for heat (diesel)		-81.6%	CONCAWE, 2008*
	France	Wheat	DDGS (43)	Straw 86% (IF)		-60%	ADEME/DIREM, 2002
			Straw 14% (96)				
		Sugar beet	Sugar, pulps (80)	Crop residues, molasses, slop (IF)		-52%	
				DDGS as animal feed (feed		-14.3%	
				wheat and soya meal) +			
				straw (IF)			
				DDGS as fuel + straw (IF)		-28.6%	
		Wheat		DDGS as feed (idem) + straw		-62.7%	
				CHF			
	EU			DDGS as fuel + straw CHP		-77%	JRC/EUCAR/
				DDGS to biogas (fuel)		-60%	CUNCAWE, 2000
				Pulps to animal feed (soya		-47%	
				meal)			
		Sugar beet		Pulps to animal feed (idem) +		-65%	
				slops to biogas (fuel)			
				Pulps + slops to biogas/heat		-80.5%	
				(idem)			
							(continued)

Table 4 (continu	(per						
			Co-product handling				
Biofuels	Regions	Feedstock	Mass allocation (% applied to biofuels)	System expansion (substitutes)	Energy allocation (% applied to biofuels)	GHG/MJ compared with fossil fuels	References
	USA	Maize		DDGS (soybean meal, corn for cattle feed) Corn gluten, meal and feed, corn oil (whole corn, nitrogen-in-urea, sov oil)	15% of net energy allocated to fossil fuel co-products		Farrell et al., 2006
Biodiesel	France	Rapeseed RME	Meal (46), acid oils (97), glvcerol (88)	Crop residues (IF)		-70%	ADEME/DIREM, 2002
		Sunflower SME	Meal (49), acid oils (97), glvcerol (88)	Crop residues (IF)		-75%	
Biodiesel	EU	Rapeseed RME		Glycerine to chemicals (propylene glycol) + meal to animal feed (soya meal from imported soybeanc)		-45.5%	JRC/EUCAR/ CONCAWE (2008)
				Glycerine and meal to animal feed (soya meal from imported		-38.8%	
				Glycerine and cake to biogas (fuel)		-63%	

			ADEME/DIREM, 2002		Wu et al., 2006		Farrell et al., 2006
-66%	59%	-80%	-77.5%	-83%	-86%		-88%
					Electricity (79.6–91.2)	Electricity + chemicals	
Glycerine to chemicals (propylene glycol) + meal to animal feed (soya meal from imported	soybeans) Glycerine and meal to animal feed (soya meal from imported soybeans)	Glycerine and cake to biogas (fuel)	Crop residues (IF)	Crop residues (IF)			Electricity (local grid electricity)
			Meal (46), acid oils (97), ølvcerol (88)	Meal (49), acid oils (97), glvcerol (88)			
	Sunflower SME		Rapeseed	Sunflower	Maize stover	Forest residues	Switchgrass
			France		USA 2030		USA
			Pure vegetable oil		Cellulosic ethanol (nilots)		

### 5.3.1 Prospects for Reducing Greenhouse Gas Emissions from Biomass Production

Agriculture greenhouse gas emissions increased by 10% between 1990 and 2000 (Stern, 2006); CH<sub>4</sub> and N<sub>2</sub>O emissions alone increased by 17% between 1990 and 2005; 88% of these emissions are explained by three sources: biomass burning, enteric fermentation and soil N<sub>2</sub>O emissions (Smith et al., 2007). Considering the increase in demand for agricultural feedstock, global agricultural greenhouse gas emissions are expected to rise by almost 30% in the period to 2020, with almost two-thirds of this increase coming from Africa, Latin America and China, half of it due to the use of fertiliser on agricultural soils (Stern, 2006). World nitrogen fertiliser demand is forecast to increase at an annual rate of about 2.6% until 2012, East Europe and Asia contributing to 81.9% of this increase (FAO, 2008b). N<sub>2</sub>O emissions alone are projected to increase by 35-60% by 2030 due to increased use of fertilisers and animal manure production (FAO, 2002).

Drastic savings in agricultural greenhouse gas emissions are needed, and agricultural practices are the key to significantly reducing agricultural greenhouse gas emissions. Options for mitigating agricultural greenhouse gas emissions fall into three categories based on the underlying mechanism: (1) reducing emissions, (2) enhancing removals from the atmosphere, and (3) avoiding (or displacing) emissions (Smith et al., 2008). The global technical mitigation potential<sup>52</sup>, including all gases for the two first categories, by 2030 is estimated to be some 5.5–6 GtCO<sub>2eq</sub> yr<sup>-1</sup> (<sup>53</sup>), mainly through reduction of  $CO_2$  emissions<sup>54</sup> (89%). The economic potential would vary between 1.5 and 4.3 GtCO<sub>2eq</sub> yr<sup>-1</sup> at carbon prices from 20 up to 100 US\$.tCO $_{2eq}^{-1}$ , respectively. At the same carbon price levels, some more 0.6-16 GtCO<sub>2eq</sub> yr<sup>-1</sup> could be avoided, by substituting fossil fuels with bioenergy generating electricity, i.e. the above-mentioned third category (Smith et al., 2008).

The uncertainty on the savings due to bioenergy is especially large because the net benefit in CO<sub>2</sub> reduction from fossil CO<sub>2</sub> displacement will depend on the greenhouse gas balance over the whole bioenergy life cycle, including direct emissions during the biomass production. Therefore, production systems of biomass for energy should necessarily contribute to the two first above-mentioned categories. Table 5 presents the main measures for mitigating greenhouse gas emissions from agroecosystems. It appears that the mitigative effects of these measures on N<sub>2</sub>O emissions are mostly uncertain, due to the lack of knowledge related previously in this review. However, since N availability is the bottom line for N<sub>2</sub>O emissions, management practices that will improve the fertilisation efficiency can help reduce N<sub>2</sub>O emissions as detailed below. Other practices could further reduce the total agricultural greenhouse gases, but the trade-off between the different gases is still unclear.

#### 5.3.2 Improving Fertilisation Efficiency

Direct field emissions can be reduced by improving fertilisation efficiency, i.e. combining reduced input and increased uptake and production. The nutrient balance expresses this difference between the total quantity of nutrient inputs entering an agricultural system, and the quantity of nutrient outputs leaving the system, in terms of kilograms of nutrient surplus (deficit) per hectare of agricultural land per year. Any surplus represents potential losses of nutrients into the environment and the correlated risk of polluting soil, water and air, whereas a deficit can reveal environmental pressures such as declining soil fertility (OECD, 2008).

Focusing on nitrogen balance, the situation is quite contrasted between OECD countries and developing countries where fertilisation inputs are much lower. Nitrogen balance is in surplus in all OECD countries, whereas in Sub-Saharan Africa notably it is in deficit, as in Kenya, Mali and Ghana, for instance (OECD, 2008; Roy et al., 2003). In two-thirds of OECD countries the nitrogen surpluses decreased between the 1990s and 2000s, whereas in a few countries such as Canada, New Zealand, Portugal and the USA it increased, notably due to the rise in fertiliser use and livestock numbers. The higher use of fertiliser is in part explained by the expansion in crop production together with a shift in cropping patterns to crops requiring

<sup>&</sup>lt;sup>52</sup> Mitigation potentials for  $CO_2$  represent the net change in soil carbon pools which were derived from about 200 studies; the emission ranges for  $CH_4$  and  $N_2O$  were derived using the DAYCENT and DNDC simulation models. All estimated potentials are followed by the mention *medium agreement, low evidence.* 

 <sup>&</sup>lt;sup>53</sup> About 20% of 1990s global greenhouse gas emissions, or 5%,
 9% and 14% for the three different economic potentials.

<sup>&</sup>lt;sup>54</sup> Notably from SOC sequestration due to restoration of organic soils; 9% CH<sub>4</sub>, 2% N<sub>2</sub>O.

431

**Table 5** A list of proposed measures (also referred to in literature as "Recommended Management Practices" RMPs) for mitigating GHG emissions from agricultural ecosystems, their apparent effects on reducing emissions of individual gases (mit-

igative effect) and an estimate of scientific confidence that the proposed practice can reduce overall net emissions (Smith et al., 2008)

		Mitiga	tive effect	cts <sup>a</sup>	Net mitigation	on <sup>b</sup> (confidence)
Measure	Examples	$\overline{CO_2}$	$CH_4$	N <sub>2</sub> O	Agreement	Evidence
Cropland management	Agronomy	+		±	***	**
	Nutrient management	+		+	***	**
	Tillage/residue management	+		±	**	**
	Water management (irrigation, drainage)	±		+	*	*
	Rice management		+	$\pm$	**	**
	Agroforestry	+		±	***	*
	Set aside, land-use change (LUC)	+	+	+	***	***
Grazing land	Grazing intensity	±		$\pm$	*	*
management/pasture	Increased productivity (e.g. fertilisation)	+		±	**	*
improvement	Nutrient management	+		±	**	**
	Fire management	+		$\pm$	*	*
	Species introduction (including legumes)	+		±	*	**
Management of organic	Avoid drainage of wetlands	+	—	$\pm$	**	**
soils restoration of degraded lands	Erosion control, organic amendments, nutrient amendments	+		±	***	**
Livestock management	Improved feeding practices		+		***	***
	Specific agents and dietary additives		+		**	***
	Longer term structural and management changes and animal breeding		+		**	*
Manure/biosolid	Improved storage and handling		+	±	***	**
management	anaerobic digestion		+	$\pm$	***	*
	More efficient use as nutrient source	+		+	***	**
Bioenergy	Energy crops, solid, liquid, biogas, residues	+		±	***	**

<sup>a</sup> '+' denotes reduced emissions or enhanced removal (positive mitigative effect); '-' denotes increased emissions or suppressed removal (negative mitigative effect); '±' denotes uncertain or variable response.

<sup>b</sup> A qualitative estimate of the confidence in describing the proposed practice as a measure for reducing *net* emissions of GHGs, expressed as  $CO_2$  equivalence. 'Agreement' refers to the relative degree of agreement or consensus in the literature (the more asterisks, the higher the agreement); 'Evidence' refers to the relative amount of data in support of the proposed effect (the more asterisks, the greater the amount of evidence).

higher inputs per kg of output, such as from wheat to maize in Australia and the United States. Some countries, whose surpluses diminished, still have amongst the highest surpluses. This is the case for Korea, Japan, Belgium, Denmark and the Netherlands, for instance, which have rather low nitrogen efficiency<sup>55</sup>, between 30 and 50%, below the averages in the OECD and the EU-15, respectively, 55%–60% (OECD, 2008). Reduced nitrogen surpluses were notably correlated with the adoption of "nutrient management and environmental farm plans", and the improvement of N-use

efficiency linked to reduced inorganic fertiliser input per unit of crop output, a closed storage system, and an optimisation of the timing and spreading of manure (OECD, 2008). Indeed, nitrogen efficiency can be increased by optimising the crop's natural ability to compete with processes whereby plant available N is lost, i.e. by matching the N supply with crop demand better [e.g. optimised split application schemes and doses, foliar application, application during stem elongation or later (Recous, 2001)] and adapt it to specific risks; e.g. avoiding nitrate application in case of leaching risk (Recous, 2001), applying fertiliser below the soil surface in case of possible volatilisation, choosing to apply nitrates when nitrification is more likely to happen, or ammonium-based fertilisers when it is denitrification; for instance, during seasonal precipitations (Mosier, 1996).

<sup>&</sup>lt;sup>55</sup> Nitrogen efficiency measured as the percentage ratio of total nitrogen uptake by plants and forage (tonnes) over the total nitrogen available from fertiliser, livestock manure and other nitrogen inputs (tonnes).

Integrated fertilisation management, including the introduction of catch crops or legumes in the crop rotation to uptake or fix nitrogen, respectively, and the incorporation of crop residues or manure spreading will also influence N<sub>2</sub>O emissions. Indeed, fertilisation efficiency strategies must be developed considering both the cropping and intercropping cycles. Whereas high N inputs may be well correlated with high N<sub>2</sub>O emissions during the vegetation period, over the year unfertilised plots can also emit high quantities of N<sub>2</sub>O, notably depending on the amounts and the C/N ratio of crop residues in the soil (Kaiser et al., 1998). Organic amendments can influence N2O emissions in three ways: (1) the amount and recalcitrance of the N supply, (2) those of the C supply, and (3) local increases in the oxygen consumption (Velthof et al., 2002). Through the crop residue C/N ratio, it is possible to influence the nitrogen mineralisation-immobilisation turnover<sup>56</sup> by microorganisms (Recous, 2001; Velthof et al., 2002), which determines the evolvement of the soil N pool, including the competition for the N substrates between plants and microorganisms, and potential N2O emissions. A narrow C/N ratio and high contents of easily mineralisable N<sup>57</sup> in crop residues would favour N<sub>2</sub>O emissions (Velthof et al., 2002). While organic amendments with high N content may accentuate N<sub>2</sub>O, NH<sub>3</sub> and CH<sub>4</sub> emissions, they may also contribute to the rise in soil organic carbon, especially in the form of sta-

bilised manure and recycled organic compost, that contain a greater fraction of recalcitrant carbon than fresh green manure, i.e. fresh crop residues (Larsson et al., 1998; Lal, 2004c; Cowie et al., 2006). Moreover, there could be possible reductions of N<sub>2</sub>O and NH<sub>3</sub> emissions in the field, depending on the soil type, through digestion of the fresh green manure and slurries before application (Oenema et al., 2005). The initial N content of composts may be more determining for leaching risk and fertilising value than amendments' stability but this latter could play an interesting role in optimising fertiliser application timing and crop N recovery (Gabrielle et al., 2005).

Although nitrogen efficiency is not identically defined across the literature, authors agree that it could and should be widely improved in future (Crutzen et al., 2008; OECD, 2008; Galloway et al., 2008). In 2002–2004, nitrogen efficiency reached 70–78%, for instance, in Italy and Greece, respectively (OECD, 2008). This issue is especially important as N-intensive biofuels could cancel out any CO2 savings due to N2O and NO<sub>x</sub> emissions. As critical examples, US corn and Brazilian sugar cane production have low N efficiency; only 30% of N input ends up in sugar cane tissues (Galloway et al., 2008). An increase in the nitrogen efficiency from 40% to 60%, resulting in the assumption that 3 instead of 5% of N input would be lost as N<sub>2</sub>O over the whole nitrogen cascade, makes the maize ethanol and rapeseed biodiesel become carbon-neutral and beneficial, respectively (Crutzen et al., 2008).

A more efficient use of fertiliser would lead to direct reduction of field emissions, while at the same time it would also imply reduction of upstream industrial emissions during the fertiliser production and spreading in the fields. It could lead to a decrease in industrial Nr creation of about 15 MtN per year, i.e. 8% of total Nr created in 2005. The same amount could also be saved through improved animal management strategies (Galloway et al., 2008).

#### 5.3.3 Other Cultural Practices

Basically, recommended management practices (RMPs) aim to improve the agroecosystem productivity while maintaining or reducing the input levels. In general, the choice of resistant varieties and an optimal adaptation of crop rotations to site-specific conditions will make it possible to reduce the greenhouse gases by combining high yields and low inputs. Farming operations should be limited as far as possible, since all inputs also imply an environmental cost starting with fossil fuel  $CO_2$ . In some cases though, the final greenhouse gas benefits will depend on the balanced gain in CO<sub>2</sub> sequestration due to enhanced biomass productivity over the cost in CO<sub>2</sub> emitted by the operations of irrigation, drainage or tillage, etc. and in other greenhouse gases. Reducing fallow will, for instance, imply higher energy, but this cost appears to be globally offset by greater benefits (Grant et al., 2004; Lal, 2004a). While some energy inputs are unavoidable, improved energy efficiency in agriculture could deliver an additional 0.77  $GtCO_{2eq}$  yr<sup>-1</sup> mitigation potential by 2030 (Smith et al., 2008).

<sup>&</sup>lt;sup>56</sup> Through mineralisation N is made available for the plants, through immobilisation/organisation N is consumed for the development of the microorganisms.

<sup>&</sup>lt;sup>57</sup> Easily mineralisable N is usually more abundant in fresh green material than in straw (Velthof et al., 2002).

The opportunity to save energy input by reducing soil tillage is the major factor that has first fostered the change from conventional tillage to reduced/conservation tillage or no-tillage. In 1999, the worldwide area under no-tillage was approximately 50 Mha, representing 3.5% of total arable land (Smith et al., 2007). Conversion from conventional tillage to no-tillage can reduce emission of farming operations by 110 to 130 kg  $CO_2$  ha<sup>-1</sup> per season (Lal, 2004b). Since soil disturbance caused by soil tillage enhances SOC losses through decomposition and erosion (see part 5.2), reduced or no-tillage often also results in SOC gain besides fossil CO<sub>2</sub> savings (Robertson et al., 2000; Lal, 2004a; Seguin et al., 2007; Smith et al., 2008). Such practices are, however, frequently combined with periodical tillage, which reverses the SOC storage trend, thus making the assessment of the greenhouse gas uncertain (Smith et al., 2007). SOC sequestration through reduced soil tillage is an explicit illustration of greenhouse gas trade-offs, or hidden costs (Lal, 2004c) that are likely to obscure the real impact of a mitigation measure. Indeed, while reduced tillage may imply SOC sequestration and globally less CO<sub>2</sub> emissions, it adversely can lead to higher emissions of N<sub>2</sub>O and CH<sub>4</sub>, though not always (Robertson et al., 2000; Six et al., 2004; Lal, 2004b; Grant et al., 2004; Chatskikh and Olesen, 2007; Oorts et al., 2007). Enhanced CH<sub>4</sub> uptake in no-tillage systems has also been reported due to higher SOC stock and the presence of ecological niches for methanotrophic bacteria (Six et al., 2004; Lal, 2004b). No-tillage can increase N<sub>2</sub>O and CH<sub>4</sub> fluxes because of higher bulk density and reduced porosity that diminish gas diffusion and increase water conservation at the surface, thereby increasing the likelihood of anaerobic conditions (Gregorich et al., 2006; Germon et al., 2007; Ball et al., 2008). Accumulation of organic matter and residue mulching at the surfaces of no-tilled fields can also favour N<sub>2</sub>O emissions (Jørgensen et al., 1997; Ball et al., 2008). However, residue mulch can also limit N<sub>2</sub>O emissions during freezing-thawing cycles by maintaining a warmer temperature that decreases the frequency or intensity of freezing events (Wagner-Riddle et al., 2007). Moreover, when N is a limiting factor, N<sub>2</sub>O emissions may be enhanced by soil tillage due to an easier diffusion through the soil matrix without being further reduced (Gregorich et al., 2006; Chatskikh and Olesen, 2007).

The determinism of N<sub>2</sub>O being especially complex, all changes related to the soil tillage system may influence the N<sub>2</sub>O emissions. The tillage timing, as well as the cumulative effect of a tillage system in the long term, will also be determinant. Furthermore, drywet or freezing-thawing cycles can create cracks and enhance the sensitivity of compacted zones to fragmentation during tillage. Thus, seedbed preparation in spring will be more efficient in reducing the proportion of compacted zones, whereas seed bed preparation in autumn will only depend on the initial state (Boizard et al., 2002). Impacts of these weather cycles may also explain how the difference in N2O fluxes between no-tillage and conventional tillage could change over time. In a review of 44 data sets, the higher N<sub>2</sub>O fluxes trend in no-tillage systems compared with conventional tillage systems was reversed after 20 years in humid climates and fluxes became similar between tillage systems in dry climates (Six et al., 2004).

Finally, no-tillage could lead to other "hidden costs" due to possibly increasing use of herbicides and pesticides (Lal, 2004b) and decreasing yields (Chatskikh and Olesen, 2007). Indeed, tilling the soil just before sowing increases soil temperature and can favour germination (Richard and Cellier, 1998). Lower N uptake in no-tilled fields could result in higher gaseous losses, and differences between greenhouse gas balances of conventional tillage and no-tillage systems eventually further shrink (Chatskikh and Olesen, 2007).

Management practices that will reduce agricultural greenhouse gases can hardly satisfy all criteria, especially as determinisms for the diverse greenhouse gas emissions can be opposed, e.g. aerobic or anaerobic conditions that enhance CO2 or N2O production, respectively; draining rice paddy fields in order to reduce the emissions of CH<sub>4</sub> enhances N<sub>2</sub>O emissions, etc. (Duxbury and Mosier, 1993). Moreover, trying to reduce N<sub>2</sub>O emissions by preventing the optimum conditions from occurring could lead to compensating for N<sub>2</sub>O emissions by decreasing the rate of reduction into N<sub>2</sub>, while complete denitrification would be the less polluting pathway to close the N cycle (Galloway et al., 2003). As long as our understanding of all involved processes remains too incomplete, the best options to reduce the agricultural greenhouse gases are to improve its overall efficiency and to reduce all inputs, especially fertilisers. Perennials are therefore particularly interesting.

### 6 The Quantitative Potential of Biofuels

To avoid  $CO_2$  emissions, substituting coal is at present a very effective way of using biomass. In the future, though, using biomass for transport fuels will gradually become more attractive from a  $CO_2$  mitigation perspective because of the lower greenhouse gas emissions for producing second generation biofuels and because electricity production on average is expected to become less carbon-intensive due to increased use of wind energy, photovoltaic and other solar-based power sources, carbon capture and storage technology, nuclear energy, and fuel shift from coal to natural gas (IEA Bioenergy, 2007). In this context, land and biomass availability will become the primary limiting factors and the ceiling for contribution to global primary energy can already be foreseen.

## 6.1 Biomass Availability: Bottom-up Approach

Biomass currently provides an annual amount of energy ranging from 40 EJ yr<sup>-1</sup> (Parikka, 2004) to 45  $\pm$  10 EJ yr<sup>-1</sup> (IEA Bioenergy, 2007), of which roughly 7 EJ yr<sup>-1</sup> are considered as modern biomass in opposition to traditional use of woodfuel (UNDP, 2000). In 2050, the total primary energy demand will vary between 800 EJ yr<sup>-1</sup> and 1400 EJ yr<sup>-1</sup> (IEA Bioenergy, 2007), and the share of biomass to meet that need is quite speculative. Due to the complexity of the numerous factors interacting to determine the potential and cost of bioenergy production, projecting future bioenergy consumption cannot be done by matching demand and supply. On the contrary, studies focus either on the supply-driven potential, i.e. resource assessment, or on demand-driven amounts required to meet exogenous targets without specifically defining the exploitable resources (Berndes et al., 2003).

Therefore, assumptions vary widely among studies and significantly impact resulting global potentials. Biomass supply could amount to 200–400 EJ yr<sup>-1</sup> by 2050, i.e. 14% to 50% of total primary energy demand, without jeopardising the world's future food supply. Considering expected average conversion efficiencies, this primary bioenergy could correspond to 130–260 EJ yr<sup>-1</sup> biofuels or 100–200 EJ yr<sup>-1</sup> electricity (IEA Bioenergy, 2007). These future bioenergy potential estimates, which are rather larger than potentials for the current situation, are based on the assumptions of future higher yields, notably thanks to perennials and advanced conversion technologies, but also through an improvement in agricultural system efficiency and the use of marginal and degraded lands. They are average values of extreme supply scenarios in 2050, from a scenario with bioenergy exclusively from waste biomass (40 EJ yr<sup>-1</sup>) up to a scenario with an intensive dedicated agriculture concentrated on the better quality soils (1100 EJ yr<sup>-1</sup>). This wide range calls for scrutiny when examining potential assessments.

#### 6.1.1 Bottom-up Models

Supply-driven studies have proven, though, that technical biomass potential could meet the amount levels of bioenergy use reported in demand-driven studies. Most studies also agree on the fact that energy crops represent the main potential biomass source compared with forest products or residues; land availability for energy crops and biomass yields hence appears to be the main key assumption (Berndes et al., 2003; Smeets et al., 2007). Diverging assumptions on the energy crop yields alone lead to a 40% difference in the maximum bioenergy potential produced from woody energy crops on about the same amount of surplus land when comparing the studies Hoogwijk et al. (2003) with Smeets et al. (2007).

Most complex approaches use models, such as IIASA's BLS model or IMAGE 2.2 model, etc., to simulate land uses and biomass availability on a geographical grid, taking into account geo-climatic conditions, the types of soils and crops, and agricultural practices. Still, the definitions of land-use patterns, geographical aggregation, and further assumptions on the evolution of both the crop yields and the efficiency of agricultural systems differ, inducing a wider range of bioenergy potentials. For instance, the management factor defined as the gap between theoretically feasible and actual crop yields, i.e. introducing yield limitation by less than optimal management practices and technologies, varies between 0.7 and 1.5 when comparing the two above-mentioned studies. A management factor above 1 expresses an increase in the harvest index, the development of irrigation and biotechnologies (Hoogwijk et al., 2005), reflecting a growing confidence in future biomass production systems up to very optimistic scenarios above current theoretical optimal systems.

					• • •		
Into: From:	Forest	Woodland/ Grassland	Cropland	Urban area*	Losses (% yr <sup>-1</sup> )	Gains (% yr <sup>-1</sup> )	Net change (% yr <sup>-1</sup> )
Forest	3.97	0.003	0.0098	0.0002	-0.33	0.14	-0.18
Woodland/	0.0014	3.44	0.001	0.0002	-0.08	0.15	0.07
Grassland							
Cropland	0.0043	0.002	1.51	0.0016	-0.52	0.70	0.19
Urban area*	n.s.	n.s.	n.s.	0.038	-0.00	5	5

**Table 6** Global land uses (Gha) and annual conversions in 2000 (Gha  $yr^{-1}$ ) (Holmgren, 2006)

n.s. not significant; \*urban area only encompasses here areas with more than 100 000 inhabitants, total area of 13.4 Gha including "other lands" for which no significant changes were identified.

Land availability for energy crop depends on the competitive uses of land for food and feed, biomaterials, forest, conservation areas, and build-up. The background assumption in most studies is that land demand for food and feed production has to be fulfilled before land is allocated to bioenergy production. Future land demand for food and feed is then assessed, taking into account expected population growth and diet evolvement; diets having a dual effect on land use. Indeed, diet in MJ day<sup>-1</sup> per capita increases as a function of the income in absolute quantity terms as well as in qualitative ones, tending to an increasing share of livestock and oilseed products in the global average diet. Growth in meat and dairy product production and consumption is expected to continue, especially in developing countries, where people eat only about 30 kg of meat per capita a year, whereas this rate is above 80 kg  $yr^{-1}$  in the industrial world. Experts predict that by 2050, nearly twice as much meat will be produced as today. The impact on land use will be severe as animal husbandry is very land-consuming. In 2002, more than 70% of the agricultural lands worldwide were dedicated to the production of animal products, while these only accounted for some one-fifth of the total calorie intake (FAO, 2003). Therefore, an analysis of the sensitivity of land availability to the evolvement of animal production systems is also necessary.

Nowadays, of the 13.4 Gha of land area in the terrestrial biosphere (Holmgren, 2006):

- 5 Gha are used for agriculture

(roughly 1.3 Gha food crops + 0.2 Gha fodder crops + 3.5 Gha pastures)

 4 Gha are under forest cover (56% subtropical and tropical forests, overall 95% are natural forests while the remaining 5% are plantation forests)  - 4.4 Gha of the rest of the land encompass semi-natural vegetation types such as savannas, etc., barren land and about 0.26 Gha of build-up area (FAO, 2002, 2003; Hoogwijk et al., 2005; Smeets et al., 2007).

Table 6 shows land-use change patterns in 2000 based on annual changes between 1990 and 2005.

Table 7 gives an overview of past and future trends of population, average calorie intakes and agricultural production. World population growth and demand for agricultural products have been slowing down since the late 1970s. Indeed, although the world average calorie intake has been rising, especially in developing countries where incomes have increased, high levels of food consumption have now been reached in many countries. In particular, China has already passed its phase of rapid growth. In the past four decades, rising yields accounted for about 70% of the increase in crop production in the developing countries and yield growth, even if not as rapid as in the past, will continue to play the same role for the next 30 years. The contribution of irrigation to this yield growth and intensification of livestock production is also expected to increase. Hence, the expansion of agricultural land at the expense of forest is expected to be concentrated in the developing countries and limited to an overall 12.5% increase, i.e. half of the increase between the early 1960s and late 1990s. However, more than half the land that could be opened up is in just seven countries of tropical Latin America and Sub-Saharan Africa, where 80% of land expansion is expected to take place, whereas other regions face a shortage of suitable land, e.g. in South Asia more than 94% of suitable land is already farmed.

Agricultural production can grow in line with demand, provided that regional shortages, notably in livestock products in developing countries, are reduced through international trade efforts. Net cereal imports

Table 7         World population and agricultural production pi	rojections at a glance (FAO, 2002)					
	Time frame	1979-1981	1997-1999	2015	2030	2050
Population in billions	World	4.430	5.900	7.207	8.270	9.322
(annual growth rate %)		(1.6)	(1.5)	(1.2)	(0.0)	(0.6)
	Developing countries:	3.259	4.595	5.858	6.910	7.987
Remaining countries	1	(1.9)	(1.7)	(1.4)	(1.1)	(0.7)
	Industrial countries:	0.789	0.892	0.951	0.979	0.986
26 countries: Western Europe, USA, Canada,		(0.7)	(0.7)	(0.4)	(0.2)	(0.0)
Australia, New Zealand, Japan, South Africa, Israel						
	Transition countries:	0.382	0.413	0.398	0.381	0.349
28 countries: Eastern Europe, former Yugoslavia SFR, CIS, Baltic States		(0.5)	(0.1)	(-0.2)	(-0.3)	(-0.4)
Calorie consumption in kcal capita <sup>-1</sup> day <sup>-1</sup> /in MJ capit	$a^{-1}$ day <sup>-1</sup> (1 kcal = 4.186 × 10 <sup>-1</sup> %) – fealorise from animal wordw	- <sup>3</sup> MJ)				
amman growm in acmand tor agricultural products in	NOT = [CalOLICS ITOLII allIIIIal production with the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the matrix of the mat	7 557/10	2 803/11	2 040/12	3 050/17	3 302/13
	WOILU AVCIAGO	7 JJ2LLU	11/C00 7	3 340114	2110C0 C	CT 170C C
		(2.1)	(2.0)	(1.6)	(1.4)	[550]
	Developing countries	2 312/9.7	2 681/11	2 850/11	2 980/12	3 236/13
		(3.7)	2	6	5	5
			(4.0)	(2.2)	(1.7)	[549]
Calorie consumption in kcal capita <sup><math>-1</math></sup> day <sup><math>-1</math></sup> /in MJ.capit	$a^{-1}$ day <sup>-1</sup> (1 kcal = 4.186 × 10 <sup>-1</sup> Industrial countries	- <sup>3</sup> MJ) 3 135/13	3 380/14	3 440/14	3 500/14	3 448/14
		1	1	4	7	4
		(1.0)	(1.0)	(0.7)	(0.6)	[1 054]
	Transition countries	3 389/14	2 906/12	3 060/12	3 180/13	3 629/15
		2	2	8	en i	2
		(-1.7)	(-4.4)	(0.5)	(0.4)	[941]
Total arable land in Gha (irrigated land in Gha)						
	WORIG	1.608 (210)	(271)			
	Developing countries	× 1	0.956 (202)	1.017 (221)	1.076 (242)	I

	Industrial countries	Ι	0.387	Ι	I	
,	Transition countries	I	(25) (25)	I	I	
Cropiand in Gna in developing countries (yields tons/na)	Wheat	0.096	0.111	0.113	0.118	I
	Rice (paddy)	(1.6) 0.138	(2.5) 0.157	(3.1) 0.162	(3.5) $0.164$	I
	Maize	(2.7) 0.076	(3.6) 0.097	(4.2) 0.118	(4.7) 0.136	I
	All cereals	(2.0) 0.408	(2.8) 0.465	(3.4) 0.497	(4.0) 0.528	I
		(1.9)	(2.6)	(3.2)	(3.6)	
	% of total harvested land	60	55	53	51	I
Forests in billion ha (annual change in Gha)						
	World	I	3.870	I	>2.940	I
			(-0.0094)		(if <-0.0094)	
	Tropical and	I	2.168	I		Ι
	sub-tropical		(-0.0123)			
	Non-tropical	I	1.702	I		I
			(+0.0029)			
	Build–up land	I	0.26	I	$0.39-0.52^{*}$	I
	worldwide in Gha					
<i>Calculations from Smeets et al. (2007) with the data from F</i> * Data from the Global Environment Outlook, UNEP 2002	AO, 2002 are in italics. in Hoogwijk et al. (2005).					

by developing countries will almost triple over the next 30 years while their net meat imports might even increase by a factor of almost five. By 2030 about 440 million undernourished people may still remain. The FAO analysed that developing countries' farmers could gain a lot from lower trade barriers, provided domestic policies permit one to remove the domestic bias against agriculture and to improve productivity and product quality to the standards demanded abroad. Investments in transportation and communication facilities, upgraded production infrastructure, and improved marketing, storage and processing facilities could be particularly important. In resource-rich but otherwise poor countries, a more export-oriented agriculture could provide an effective means to fight rural poverty (FAO, 2002).

#### 6.1.2 Availability of Agricultural Land

Estimation of land availability for energy crops finally essentially relies on the surplus land area that may be released in certain regions either because of higher yields and intensified production, or some agricultural lands being abandoned as they become no longer suitable for food and feed production.

First expectations are reinforced by the fact that in many countries average wheat yields, for instance, for the period 1996-2000 were below the agroecologically attainable levels; in India, Brazil, and even Australia and the US, they were calculated to be roughly half the maximum levels (FAO, 2002 in Smeets et al., 2007). Also, feed conversion efficiency<sup>58</sup> of bovine meat and dairy products was in 1998 three to four times higher in industrial countries than in developing countries, indicating that part of the land demand for livestock production could be outpaced by an increase in efficiency of livestock production systems. Moreover, the intensification of husbandry systems could permit one to spare grassland for other crops as some part of suitable croplands are currently used as pastures, especially in developing countries (Smeets et al., 2007).

Testing diverse scenarios of combined population growth, and change in diets and food and feed producC. Bessou et al.

tion systems (Table 8), recent studies show that considerable parts of agricultural lands could be allocated to bioenergy production without jeopardising the food and feed supply: 0.15 to 2.4 Gha (Hoogwijk et al., 2003), 0.6 to 1.3 Gha (Hoogwijk et al., 2005) and 0.7 to 3.6 Gha (Smeets et al., 2007). The highest estimates, notably in the systems 3 and 4 of Smeets et al. (2007), are rather too optimistic though, combining very intensive agricultural production with high yield increases and landless animal production, together with a management factor of 1.5, that indicates even further possible improvements; more irrigation, more fertilisers, while system 4 already corresponds to a 25% addition to the yield levels at a very high rain-fed/irrigated level of agricultural technology. The landless animal production system especially leads to consequent surplus pasture areas, up to 613-820 Mha in some developing countries, which can partly serve for bioenergy production (Smeets et al., 2007). However, such a system could hardly be implemented in these regions by 2050. On the contrary, a low productive agricultural system (Y1 in Table 8 = "Low External Input" in the study Hoogwijk et al., 2003), which could be roughly compared with an organic production system, does not permit one to grow any energy crop on agricultural land. This scenario suggests yields in 2050 that would be around the same as today's, once again emphasising that agricultural intensification is a prerequisite.

Considering medium population scenarios with affluent diets and relatively high production systems across the three studies, surplus agricultural land areas allocated to energy crops would vary between 0.45 and 1.3 Gha. Maximum bioenergy potential on these surplus agricultural land areas would vary between 135 EJ yr<sup>-1</sup> and 409 EJ yr<sup>-1</sup> (LHV = 15 GJ tDM<sup>-1</sup>), i.e. roughly 25% to 50% of total primary energy demand in 2050, considering a medium energy demand scenario of 837 EJ yr<sup>-1</sup>. Nevertheless, this maximum potential is simulated considering a unique type of biomass source which is short rotation coppices (SRC). These crops provide high yields and are convenient for assessing maximum primary bioenergy potential based on gross energy contents, whereas potentials expressed based on first generation biofuel crops would not be as relevant without calculating conversion factors. However, SRC are not representative of the portfolio of energy crops and present higher yields than other energy crops. Considering the cereal yields of the High External Input system and assuming that the

<sup>&</sup>lt;sup>58</sup> Feed conversion efficiency is defined as the amount of animal product produced per amount of animal feed input.

(continued)										
	(3-4%), extensive grassiand areas, rest land areas sensitive to diverse ecological stresses such as scarce water resources: 50% (A1, A2) 90% (B1, B2)	$MF^{\bullet} = 0.78$ $MF^{\bullet} = 0.82$								
$\begin{bmatrix} *max & yx \\ max & yx \end{bmatrix}$ $\begin{bmatrix} +max & yx^{-1} \\ max & yx^{-1} \end{bmatrix}$ LHV = 19 GJ t <sup>-1</sup>	production, forest areas, tundra, nature reserves: 5% (A1, A2), 15% (B1, B2), urbanisation	Regional yields of 12 food crops weighted by MF	13.2	12.8	12.5	11.3	9.4	8.7	-excluded and allocated areas	structure the IPCC SRES: A1, A2, B1, B2
*Yields weighted by MF = $6-34 \text{ tDM}$ ha <sup>-1</sup> yr <sup>-1</sup>	Land-claim exclusion factors (% of global land area): agricultural land needed for food and feed	A2 A1 B2 B1	A1 B1	B2	A2	A2	B2	A1 B1	13	Hoogwijk et al., 2005 interpreting
$P_1,D_3,Y_2 => 0.3 (504)$ $P_2,D_3,Y_2 => 0.45$ $(171)$ $P_3,D_3,Y_2 => 0.15 (57)$	Management factor (MF) for energy cr ops: 0.7								+ 3.5 grasslands	
Y1,D2–D3,P => 0 P1,D2,Y2 => 2.6 (988) P2,D2,Y2 => 2.4 (912) P3,D2,Y2 => 1.9 (722) P1,D3,Y2 => 0.8 (304)	is doubled to take into account losses, risks, uneven accessibility to the resources, etc.			i					20% irrigated)	
= 20 tDM ha <sup>-1</sup> yr <sup>-1</sup> HHV* = $19 \text{ GJ t}^{-1}$ D1 "not likely"	constant = 5 Gna No deforestation for bioenergy production Area manded for food and feed	2.2 5.9	11.5	10.1	10.1	11.3	9.4	8.7	15 cronlands	et al., 2003
Yields weighted by MF	Total agricultural land is	Y1 Y2	D3	D2	D1	P3	P2	P1	5=	Hoogwijk
Available surplus agricultural area in Gha (EJ yr <sup>-1</sup> of primary energy from dedicated crops)	Constraints	Production systems: yields in tDM gr <sub>eq</sub> ha <sup>-1</sup> yr <sup>-1</sup> (irrigated)	son in grain <sub>eq</sub>	per per lay <sup>-1</sup> ( day <sup>-1</sup>	Diet MJ c kg <sup>-</sup> kg	lation bitants)	d popu on inha	Worl (billi	Land available pool (Gha)	Scenarios for 2050

Table 8 (cont	inued)								
Scenarios for 2050	Land available pool (Gha)	World population (billion inhabitants)	Diet per person in MJ day <sup>-1</sup> (grain <sub>eq</sub> kg <sup>-1</sup> day <sup>-1</sup> )	Production systems: yields in tDM greq ha <sup>-1</sup> yr <sup>-1</sup> (irrigated)	Constrain	ts			Available surplus agricultural area in Gha (EJ yr <sup>-1</sup> of primary energy from dedicated crops)
					Impact o change Pressure not inc Managen energy B1: 1.2	f climate e is incluc on land q cluded nent Fact r crops: A 3; A1: 1.5	led luality is or (MF) f .2, B2: 1.	it of	A1 => 1.3 (409) [20] B1 => *1.3 (398) [20] A2 => 0.6 (129) [16] B2 => *1 (279) [18]
Smeets et al., 2007	13	8.	13	Systems Animal production	S <sub>1</sub> mixed	S <sub>2</sub> mixed	S <sub>3</sub> landless	S4	Yields weighted by MF(=1.5) = 4-39 tDM ha <sup>-1</sup> yr <sup>-1</sup> [mean yields tDM ha <sup>-1</sup> vr <sup>-1</sup> ]
	-excluded and allocated areas		details by regions and animal products contribution	Feed conversion efficiency Technology for crop production	high very high	high very high	high very high	high super high	$HHV = 19 \text{ GJ t}^{-1}$
			given by the authors	Irrigation Average yield increases (19 crops) between 1998–2050	none _   ×2.9	partly ×3.6	partly ×3.6	partly ×4.6	S1 => 0.7 (215) [16] S2 => 1.15 (455) [21] S3 => 3.3 (1,101) [17] S4 => 3.6 (1,272) [20]

increase in demand for feed from grasses and fodder compared
with 1998, is added to total demand for food crops. Allocated
land areas excluded from available pool for bioenergy crops:
other land including uncultivated land, barren land, etc.,
build-up, plantations and natural forests, permanent crops,
crops not in the model (13% of total harvested area).
agricultural land needed for food and feed production including
trade at regional levels to ensure food and feed security.
No deforestation for bioenergy production
Water is excluded as a limiting factor except in arid and semi-arid regions
Irrigation is limited to areas in which climate, soil and terrain permit it
Impact of climate change is excluded
Management Factor (MF) for energy crops: 1.5
s are the closest to FAO simulations of business-as-usual scenario for 2050. "not likely" expresses the opinion of this article's authors and not any mentioned co
nts' authors; *H/LHV: high/low heating value i.e. including/or not the energy recovered from vapour condensation; •Average management factors that aff
food crons calculated with IMAGE 2.2 model at a concambical arid cell level of 0.5° × 0.5°. *read on the graphs

ent une grapus. ۲ 2 à 5 gougrapi Ħ Figures in italics a of the assessment yields of the 12 fc

The yields of the 19 crops are calculated with IIASA model and weighted (0–100%) in function of the suitability (5 levels) of the land area allocated. Demand for feed crops, including

whole cereal crop is harvested for energy purposes, the 0.45–1.3 Gha would only provide 40-279 EJ yr<sup>-1</sup> (LHV = 15 GJ tDM<sup>-1</sup>) (Hoogwijk et al., 2003). Yields of energy crops assumed in other studies range from 7 to 49 tDM ha<sup>-1</sup> yr<sup>-1</sup> (Smeets et al., 2007). Moreover, SRC production systems require specific machines and overall high investments that are unlikely to be widely affordable in order to reach optimum yields all over the world by 2050.

Theses studies are often quoted throughout the literature, because of the scarcity of such global assessments. A deep insight into these studies' background assumptions would make it possible to apprehend the relevance of the results better. However, other assessments based on different models and hypotheses would be needed to test the robustness of the results.

#### 6.1.3 Biomass from Forest and Residues

Studies also differ fundamentally in their conclusions about the availability of forest biomass for bioenergy purposes (Berndes et al., 2003). Assumptions diverge both concerning the projected forest plantations and forest growth rates, and the volume restrictions due to competitive wood industrial demand. According to Smeets et al., 2007, energy potential from surplus forest growth in 2050 ranges between 59 EJ yr<sup>-1</sup> in the case of a low plantation scenario and a high forest product demand, and 103 EJ yr<sup>-1</sup> in the case of a high plantation scenario and a low demand; meanwhile, it could reach 74 EJ yr<sup>-1</sup> in a medium scenario. Woodfuel is one of the main forest products; about 60% of the world's total wood removals from forests and trees outside forests are used for energy purposes. Demand for woodfuel will remain strong for many years to come, although its share in total energy demand is expected to decrease, as most developing countries have adopted energy policies aimed to promote the use of other options by households, such as liquefied petroleum gas (LPG), bottled gas and kerosene. This decrease is being largely compensated for, though, by the increased woodfuel use for industrial energy in developed countries (Trossero, 2002). The overall demand for forest products will continue to grow as world population and income grow, but improvement in forest exploitation efficiency, increases in plantation and an expansion of the role of trees outside forests should ensure sufficient wood supply.

The key questions now are where it should come from and how it should be produced (FAO, 2002). Ensuring a sustainable woodfuel supply for the poorest people of developing countries remains a serious problem. In places with intensive woodfuel use, for example around large urban centres and in zones with a high concentration of commercial activities such as brickmaking, the pressure on woodfuel supply sources can be heavy, with consequent deforestation. Therefore, generalisation on sustainability of woodfuel use at the local level cannot be done without careful analyses in the field (Trossero, 2002). Because of the decentralised nature of wood energy systems, energy and forestry statistics seldom include the same level of detail about woodfuel consumption as for other conventional energy sources or forest products (Trossero, 2002). Some studies hence only mention forest biomass potential from residues.

Indeed, forest exploitation produces high amounts of residues, although the production of wood-based materials is continually increasing in efficiency, i.e. the ratio of residues over final products is decreasing (FAO, 2002). About 60% of the total harvested tree is left in the forest and the non-commercial species are often felled and left on site to ease the logging, providing a valuable source of bioenergy (Parikka, 2004), especially as industrial round wood production is expected to rise by 60% by 2030, from current levels to around 2400 million m<sup>3</sup>, with one-third from plantations, about twice as much as today's plantation production of 400 million m<sup>3</sup> (FAO, 2002). After processing, about half of the log input becomes wastes that can have various alternative uses such as chips for pulp or chipboard, etc., and fuel for internal energy use or as compacted commercialised fuels, depending on the sale values on the diverse markets (Parikka, 2004).

Overall, biomass residues may be classified as follows: primary ones from agriculture (crop residues) or forest (logging residues), secondary ones from agriculture (from food processing, animal manure) and forest (mill and manufacture residues), and tertiary ones including all kind of final biomass waste. Although residues can provide a substantial source of biomass in a global energy context (Berndes et al., 2003), potential assessments are highly uncertain. First, residue generation is a multiplier factor of assumed total food and feed production. Fischer and Schrattenholzer (2001) thus stressed that this factor will furthermore decrease while the harvest index of crops is expected to increase

thanks to progress in agricultural technology (e.g. bioengineering, agricultural practices, etc.), which implies another degree of uncertainty. Second, only a fraction of total residues is recoverable in practice. Third, residue amounts available for energy production also depend on possible alternative uses of residues such as demand for feed, fertiliser (including the need to maintain soil quality), and for wood products, although these last may also partly become an eventual energy source. Most studies assume a recoverability fraction of 25% for primary residues, and higher fractions for secondary and tertiary residues between 75 and 100%. Estimates, from the literature, of potential contribution from biomass residues in 2050 vary between 38 and 245 EJ yr<sup>-1</sup> (Hoogwijk et al., 2005). In Smeets et al. (2007), this range is reduced to 76 to 96 EJ yr<sup>-1</sup> as residues needed for feed are excluded. These amounts should be further reduced though when compiling all residue demands at regional levels. Considering recoverability limits and some competing uses, the total maximum available residues for bioenergy accounts for roughly 6% of total bioenergy potential (Hoogwijk et al., 2003; Smeets et al., 2007).

#### 6.1.4 Geographical Distribution

Studies mostly agree on the fact that maximum bioenergy potentials, as a function of surplus nonfood competitive land areas, are likely to be concentrated in a few regions. In Sub-Saharan Africa, Oceania, the Caribbean and Latin America, large areas suitable for crop production are currently used as pastures (Smeets et al., 2007). Intensification of animal production systems and overall yield increase in these regions would lead to the release of land areas for energy crops. Together with the CIS and Baltic States, these regions have the highest bioenergy potentials from energy crops on surplus agricultural lands (Hoogwijk et al., 2005; Smeets et al., 2007). The largest energy potential from surplus forest growth is found in the CIS and the Baltic States, the Caribbean and Latin America, and partially North America and Western Europe Smeets et al., 2007. Figure 19 shows regional bioenergy future potentials of the four scenarios in Smeets et al. (2007). Figure 20 illustrates that considering current technologies and land and water availability as well as food insecurity, biofuel potential is



**Fig. 19** Total technical bioenergy production potential in 2050 based on systems 1-4 (EJ yr<sup>-1</sup>; the left bar is system 1, the right bar is system 4) (Smeets et al., 2007)



**Fig. 20** Biofuel production potential (Von Braun, 2007). Sources: data on food insecurity are from FAO (2006). The land availability index is derived by the author based on data from

nowadays still limited in Africa and Asia notably (Von Braun, 2007).

Oceania is the least land-stressed region, whereas Japan is the most land-stressed country. The Middle East and North Africa, and South and East Asia have relatively scarce agricultural land (Smeets et al., 2007). High land requirements for urbanisation were estimated in South and East Asia, mainly India and China, whose abandoned agricultural land areas should increase at the end of the century with the decrease in population growth (Hoogwijk et al., 2005). In industrial or transition countries, bioenergy production requires less drastic changes than in developing countries. Furthermore, in Eastern Europe, the CIS and Baltic States, food consumption and population are projected to decrease, which makes bioenergy potential in these regions more robust than in others. Regions with the highest potentials could turn into bioenergy exporters. In developing countries, bioenergy may provide new incentives for investments in agricultural research and, by providing new income, carry out a modernisation of the agricultural production systems, with a positive feedback on yields possible (Smeets et al., 2007). This is especially crucial for regions that are expected to have high bioenergy potential, such as Sub-Saharan Africa and Latin America, but which also suffer from soil erosion or nutrient depletion. In Sub-Saharan Africa, some 95 Mha of land is threatened with irreversible degradation if soil nutrient FAO (2007); IEA (2007a,b); and USDA (2006). Data on water availability are from WRI (2007)

Table 9	Expe	ected imp	pacts of	clir	nate	change
on globa	al ce	real proo	duction.	Ac	lapted	l from
Tubiello	and	Fischer	(2007)	in	Von	Braun
(2007)						

	1990-2080
Localisation	(interval change %)
World	-0.6 to -0.9
Developed countries	2.7 to 9.0
Developing countries	-3.3 to -7.2
Southeast Asia	-2.5 to -7.82
South Asia	-18.2 to -22.1
Sub-Saharan Africa	-3.9 to $-7.5$
Latin America	5.2 to 12.5

depletion continues (Henao and Baanante, 2006) in (Agard et al., 2007). Furthermore, these regions are expected to be inversely impacted by climate change. According to Tubiello and Fischer (2007), cereal production would decrease by 3.9–7.5% between 1990 and 2080 due to climatic risks, whereas it would increase by 5.2–12.5% in Latin America (Table 9).

As shown in Fig. 21, many areas are affected by degradation; Tropical Africa, Southeast Asia, North-Central Australia, Central America, the Caribbean, southeast Brazil, and boreal forests in Alaska, Canada and eastern Siberia are the most severely touched. Here, land degraded areas are defined as areas with a combined declining trend of Net Primary Production and declining Rain-Use Efficiency over the past 22 years, excluding the simple effects of drought.



Fig. 21 Global land degradation using biomass production and rain-use efficiency trends between 1981–2003 (Bai et al., 2007, online: http://www.fao.org/nr/lada/images/maps/ndvi\_preliminar.jpg)

Table 10     Tropical degraded       lands with potential for plan-       table activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activities activititities	Degraded land areas	Forest fallows	Deforested watersheds	Degraded drylands	Total
in Grainger (1991). Total	Africa	59	3	110	172
degraded land in the tropics	Asia	59	57	110	226
could average 2 Gha	Latin America	85	27	10	222
(Grainger, 1988)	Total	203	87	330	620

23.5% of the world land area is in a state of more or less severe degradation (Bai et al., 2008). The most degraded areas are mainly associated with forest degradation, although the precise history of land degradation processes has to be investigated on a regional scale. The degraded areas represent a loss of NPP of about 800 million tons of carbon that were not fixed during this period, added to  $CO_2$  emissions into the atmosphere of one or two orders of magnitude more than this amount from the loss of soil organic carbon and standing biomass (Bai et al. 2007 in Agard et al. 2007).

From about 2 Gha of degraded areas in the tropics solely, some 420 to 620 million ha, could be suitable for afforestation or vegetation enhancement, respectively, subtracting or not the forest fallows that are part of shifting cultivation systems (Table 10, Grainger, 1991). Houghton et al. (1991), also assessed that around 580 million ha of degraded land, formerly covered with forests or woodlands, may be available

to be planted or managed as plantations. Parts of these areas could be used to produce energy crops and SRC could be a means of recovering parts of the lost former sequestered carbon. Although yields on these lands will not be optimum, some species with low needs, such as Jatropha curcas, could be used in agroforestry systems (see picture below) in order to limit further degradation of these areas and CO<sub>2</sub> emissions. On 420-580 million ha of degraded land, also referred to as "low-productive land", some 8-11, and 24–33 up to 80–110 EJ yr<sup>-1</sup> could be produced considering yields of 1, 3 and 10 tDM  $ha^{-1}$  yr<sup>-1</sup>, respectively (Hoogwijk et al., 2003; Smeets et al., 2007). Finally, Smeets et al. (2007) also mentioned that some further 247 EJ  $yr^{-1}$  could be produced on the 3.6 Gha classified as other land. This is an absolute finite potential, assuming that all other land could be allocated to energy crops. "Other land" or "rest land" areas include several kinds of natural vegetation and other sites remaining once agricultural, build-up and

forest land areas are allocated that are not particularly suitable for production. They are therefore also classified as "low-productive lands". More specific studies would be necessary to evaluate on a regional scale the effective availability of rest land areas to be cultivated and the overlapping between these areas and degraded land areas.



Agro-forestry system: Jatropha curcas/chili Belize, February 2003 bessou©INRA

To summarise, biomass availability for bioenergy can be assessed through five steps, each corresponding to a gradually decreasing potential (Hoogwijk et al., 2005). The first theoretical potential in 2050, some  $3500 \text{ EJ yr}^{-1}$  (Hoogwijk et al., 2005); 4435 EJ yr<sup>-1</sup> (Smeets et al., 2007) takes into account the conversion of solar energy by vegetation (Net Primary Productivity), giving an upper limit of primary biomass energy potential on the total terrestrial surface. This indicative potential is severely reduced at a regional level by the multiple uses of land, which leads to a smaller geographical potential, also diminished due to losses through conversion processes of primary biomass to secondary energy carriers corresponding to a technical potential. Finally, economic and other socio-technical constraints may also drastically limit final economic and implementation bioenergy potentials.

Economic feasibility notably depends on raw material costs, conversion efficiency and incentives that will translate into political orientation choices. In the case study of woody biomass for energy production as a means of greenhouse gas reduction, Dornburg et al. (2007) showed that, in Poland, biomass potential is larger than the amount that could make possible cost-effective greenhouse gas savings at low costs. Biomass cost is notably strongly affected by land price elasticity; so that electricity and methanol from woody biomass remain interesting ways of cost-effective reductions in greenhouse gas emissions only as long as markets are large enough to absorb the supply without lowering market prices. In some cases, theoretical and technical potentials might be well above these thresholds. For biomaterials, market volumes are an even more critical issue. Hence economic potential highly varies depending on the supply curves of bioenergy and the internalised environmental costs, which also highly depend on market sizes.

Implementation potential, that actually defines the overall final bioenergy potential, is further limited by diverse constraints that can barely all be embraced in any assessment. For instance, the actual feasibility of the implementation of energy crops and the impacts on degraded lands are merely mentioned; studies refer to exogenous assessments of the actual extent of degraded land that could be suitable for plantation establishment (Berndes et al., 2003). Also, many environmental aspects are not taken into account, such as impacts on biodiversity, erosion, water and nutrient uses, etc. The issue of water and nutrient uses is especially crucial in SRF with species such as willow or eucalyptus that take up lots of water and nutrients from the soils. Large-scale energy crop implementation would in some countries, such as Poland or South Africa, lead to further exacerbation of an already stressed water situation (Hoogwijk et al., 2005).

Therefore, more research is needed in modelling interactions between the competitive land uses and ecological issues (Berndes et al., 2003). Notably, landuse change from forest area to bioenergy area is excluded in most studies, but land-use change from forest area to agricultural land for food and feed production is not, meaning that the actual forest areas decrease/shrink. In the scenario with high demand for food and low technology development (A2) a significant amount of forest is cut down: 45% of South American forest area could disappear within 100 years (Hoogwijk et al., 2005). Part of the abandoned agricultural land areas should therefore be allocated to reforestation and not to bioenergy production, in order to compensate for the loss in forest areas due to land clearing for agriculture.

### 6.2 Focus on Europe

A report of the European Environment Agency (EEA, 2006b) focused on how much bioenergy could be produced without harming the environment, leading to the following strict assumptions: the use of energy crops with low environmental pressure, the preservation of current protected forests and extensively cultivated agricultural areas (almost 6 million ha of grassland, olive groves and dehesas), a minimum 30% share of used agricultural area dedicated to *environmentallyoriented farming EOF*<sup>59</sup>, 3% of intensively cultivated agricultural land set-aside by 2030, ambitious waste minimisation strategies, and the further liberalisation of agricultural markets with a reduction of 40% below the greenhouse gas emission 1990 level that would make the price for carbon emission permits increase.

The results show that primary biomass potential could rise to 7.9 EJ yr<sup>-1</sup> (190 Mtoe yr<sup>-1</sup>) in 2010 (target 6.3 EJ yr<sup>-1</sup>  $\approx$  150 Mtoe yr<sup>-1</sup> by 2010) up to around 12.3 EJ yr<sup>-1</sup> (295 Mtoe yr<sup>-1</sup>) in 2030, i.e. 17% of European current global energy consumption (EEA, 2006b). This *conservative estimate*<sup>60</sup> concludes that the largest potential comes in the short term from the waste sector, especially in Germany and France (all kind of waste included: around 4.2 EJ yr<sup>-1</sup>  $\approx$  100 Mtoe yr<sup>-1</sup>), whereas energy crops from agricultural land would overtake it in the long term (up to around 5.9 EJ yr<sup>-1</sup>  $\approx$  140 Mtoe yr<sup>-1</sup>). *Environmentally-compatible* bioenergy potential<sup>61</sup> from forestry is

estimated to be almost constant throughout the period analysed (around 1.7 EJ yr<sup>-1</sup>  $\approx$  40 Mtoe yr<sup>-1</sup>) (EEA, 2006b).

The modelling of the released and set-aside land area was based on the CAPSIM model (EuroCare, 2004). The available arable land within the EU-22 that could be used for dedicated bioenergy production increases from 13 million ha in 2010 (8% of the total UAA) to 19.3 million ha in 2030 (12% of the UAA). Assumptions include that current areas used for energy crop production remain available for bioenergy production, but other drivers interact to influence this land availability. Most of the land is made available through release of land from food and fodder production. As maintaining the current European food self-sufficiency level was set as a framework condition, the competition between food and bioenergy production was assumed to be relevant only for the part of agricultural production that corresponds to projected food exports. Thus, consideration of the competition effect between bioenergy and food production was restricted to France and Germany, the only member states which are projected to combine a very high export surplus for cereals with a large agricultural land area (EEA, 2006b).

The main "suppliers" of available land for bioenergy production are Poland, Spain, Italy, the United Kingdom, Lithuania and Hungary. These countries, plus Germany and France, will produce more than 85% of the environmentally-compatible bioenergy potential in Europe. Population size and the economic competitiveness of the agricultural systems in each member state are the main factors determining land potential. On the basis of the available land and an environmental ranking of energy crops, sustainable crop mixes were determined for different environmental zones in Europe. Europe was divided into 13 zones with homogeneous pedo-geo-climatic characters. The environmentally-compatible agricultural bioenergy potential is shown in Fig. 22, taking into account the LHV for the conversion of the harvested dry biomass into an energy potential (EEA, 2006b).

Figure 22 shows that high shares of bioenergy would be supported by a complete shift to second generation biofuels with notably a crop mix, not at all

<sup>&</sup>lt;sup>59</sup> EOF encompasses organic farming and high natural value farming (NHV).

<sup>&</sup>lt;sup>60</sup> Quoted as written in the study: related to the overall value judgements in the study that limit the available potential including strict environmental assumptions.

 $<sup>^{61}</sup>$  "Environmentally-compatible" bioenergy potential = the quantity of primary biomass that is technically available for

energy generation based on the assumption that no additional pressures on biodiversity, soil and water resources are exerted compared with a development without increased bioenergy production (EEA, 2006b).



**Fig. 22** Environmentally-compatible agricultural bioenergy potential (EEA, 2006b).

Note: No data available for Cyprus, Luxembourg and Malta. 'Oil crops' comprise rapeseed and sunflower. 'Crops for ethanol' include the potential of grains from maize, wheat, barley/triticale. 'Crops for ligno-cellulosic ethanol' cover the energy value of the whole plant (corn and straw) for wheat and barley/triticale.

relying on oil crops any more. Following the scenario presented in Annex 3 of the Biomass Action Plan, the distribution of biomass used in 2010 would be: 50% for heating, 37% to produce electricity and 13% in the transport sector. The main drivers in the increase in bioenergy potential are productivity increases<sup>62</sup>, notably due to the introduction of advanced technologies, and the assumed liberalisation of the agricultural sector, notably linked to the CAP reform that should result in additional land area available for dedicated energy crops. The EEA study shows that there is a significant development potential for bioenergy within Europe, even considering strict environmental assumptions (EEA, 2006b). It is, though, in agreement with the previous detailed studies, showing that this potential will be limited by land availability issues, especially in Western Europe.

'Crops for biogas' are maize (whole plant), double cropping systems, switch grass and the grass cuttings from permanent grassland. 'Short rotation forest and perennial grasses' include poplar, willow, miscanthus, reed canary grass, giant reed and sweet sorghum, which may often be used in whole-plant conversion systems like gasification, or biomass-to-liquid processes

# 6.3 Liquid Biofuel Potential: Top-Down Approach

The top-down approach does not aim to assess maximum biomass potential but to study the feasibility of biofuel development plans, i.e. the land areas that would be necessary to produce certain amounts of biofuels (Fig. 23). Given the numerous types of feedstocks and processes that determine biofuel yields per unit of biomass input, there would be multiple scenarios for future global biofuel potentials and as many possible studies. Global assessments therefore extrapolate current biofuel trends and implemented target policies.

In 2004, a study estimated that 37% of total harvested grains, oilseeds and sugar crops area would have been needed to reach a 10% biofuel share (on an energy basis) in world transport fuel consumption within the major biofuel producing regions (Brazil, US, EU<sup>63</sup>,

 $<sup>^{62}</sup>$  Assumed yield increases: 1% per year for conventional arable crops, 1–2.5% for dedicated energy crops. A lower yield increase of 1% for all crops would reduce the bioenergy potential by 2% in 2010, and by 13% in 2030.

<sup>&</sup>lt;sup>63</sup> In both studies here (OECD, 2006) and (Fulton et al., 2004) data concern the years 2000–2004, hence projections for the



Fig. 23 Land areas (ha) needed to produce 3000 L of gasoline equivalent of biofuel energy; 1000 L per type of biomass feedstock except for Brazil 2000 L of ethanol from sugar cane. Drawn from data in OECD (2006)



**Fig. 24** Biofuel shares in transport fuel consumption and land requirements for 10% biofuel shares in major biofuel producing regions (OECD, 2006).

Notes: Current biofuel shares include ethanol and biodiesel only – shares are on an energy basis. World area shares are calculated relative to land used for cereals, oilseeds and sugar globally (World 1) and within the five major biofuel producing regions

Canada) (OECD, 2006). Except in Brazil and Poland, such a share would require excessive amounts of cropland areas (Fig. 24). In Brazil, the combination of

only (World 2). All areas requirements are calculated on the basis of average crop area and yield data for 2000–2004 and transport fuel consumption in 2004. For these calculations, the 2004 shares in the feedstock mix are assumed to remain unchanged. Note that calculations for the EU exclude ethanol transformed from wine which represented about 18% of EU ethanol production in 2004

high sugar cane yields and low transport fuel consumption per capita make it possible to go beyond 10% shares up to even more than the current 40%. In the EU, on the contrary, the situation is especially critical as land endowment per capita is particularly low. Moreover, biodiesel represents the largest share of

European Union encompass only the 15 Member States before the entrance of the other 12 members if not mentioned.

biofuels produced in Europe and also consumes much land area, about five times more than ethanol from sugar beet (OECD, 2006).

As an extrapolation of land areas allocated to biofuel production in 2004 and the correspondent transport fuel share of biofuels, 72% of the total cropland area of relevant crops harvested in 2004, would have been needed to reach the same target in the EU (OECD, 2006). Fulton et al. (2004) assessed the potential impacts on cropland areas if the US and the EU were to expand 1st generation biofuel production to reach targets of 5% in 2010 and 10% in 2020 displacement of both road transport gasoline and diesel future consumption (on an energy basis). Considering constant total cropland areas, including set-aside lands in the EU, and annual increases in crop yields (1% for all crops) and conversion efficiencies (1% for ethanol and 0.3% for biodiesel), by 2010 some 20 and 21% of total cropland area would be needed to produce enough biofuels in the EU and the US, respectively, rising to 38% and 43% by 2020. Especially for biodiesel to displace 10% of consumed transport diesel much higher land allocations would be necessary, even over 100% of projected EU oil-seed croplands as soon as by 2010 or by 2020 in the US (Fulton et al., 2004).

Despite fundamental differences between the two estimates, i.e. the time frame and calculation method, total land areas, assumptions on crop yields and fuel consumption increases, the share of sugar beet ethanol in the EU and the biodiesel production projection in the US, both estimates show very high requirement of land areas to reach the 10% biofuel target. They also emphasise that it may make sense to focus more on ethanol blending rather than on biodiesel, since in the medium term substantial amounts of biofuels can only be achieved if the feedstock mix is adjusted in favour of commodities with a higher biofuel output per hectare. This may also include imported feedstock and biofuels (OECD, 2006).

Within the EU (27 Member States) and Ukraine, a 10% biofuel share could be reached by 2020 relying on low-cost 1st generation biofuels without endangering food security or nature conservation, assuming that demand for other bioenergy sectors remains confined to forest feedstock, i.e. 100% of the agricultural feedstock for bioenergy is dedicated to biofuels, and provided that 30% of the target is met by imports from outside Europe, without imports a 9% share could be reached by 2030. Compared with previous estimates, higher

biofuel potential within the EU is notably due to higher potentials in new member states. However, if 10% is not the end term for biofuels, 2nd generation biofuels will become more competitive (REFUEL, 2008).

The development of 2nd generation biofuels produced from residues and dedicated crops appears crucial to lower the pressure on cropland areas. Compared with 1st generation biofuels, 2nd generation biofuels can be produced from a wider range of feedstock, including agricultural and wood-related residues without direct use of land, or dedicated crops that can be grown on a wider spectrum of soils. Second generation biofuels also make two- to four-fold higher land-use efficiencies possible due to higher crop yields, less agricultural management inputs and better conversion efficiency.

By 2030, a 15 to 25% biofuel share could be met by a mix of 1st and 2nd generation biofuels produced in Europe only (REFUEL, 2008). Considering the estimated cellulosic feedstock from residues and bioenergy crops on marginal lands in the US ( $\approx$ 388.5 million dry tons per year at US\$50.ton<sup>-1</sup>) and a conversion efficiency in a post-2010 scenario of 400 L ethanol per dry cellulosic feedstock ton, second generation ethanol could provide up to 26% of US annual motor gasoline consumption by 2020 without using dedicated cropland area (in Fulton et al., 2004).

Furthermore, yields of dedicated lignocellulosic energy crops are expected to grow much faster than those of conventional crops as research in breeding new varieties or adapting crop farming are still in the early stages. Finally, cooking oils and other municipal wastes could contribute to a lower extent to biofuel production, poor economics of scale being compensated for by low feedstock prices. In the EU and the US, 1 and 1.9 billion litres of biodiesel, respectively, could be produced annually, about one percent of diesel consumption in the US in 2010 (Fulton et al., 2004).

# 6.4 Projected Worldwide Biofuel Production and Consumption

In the reference scenario of the IEA with 2004 as the baseline and a global primary energy demand increase of 1.6% per year (IEA, 2006), total world



**Fig. 25** Fuel ethanol, production and projections to 2020 (Fulton et al., 2004) added production data for 2006 and 2007 from F.O. Licht



Fig. 26 Share of biofuels in road-transport fuel consumption in energy terms (IEA, 2006)

production of 1st generation biofuels is projected to climb up to 3.85 EJ yr<sup>-1</sup> (92 Mtoe) by 2030, expressing an average annual growth rate of 6.3%. Figure 25 gives an overview of past and projected ethanol production, showing that global trends follow the predicted scenario, the EU still produces less than expected, compensated for by the US where it is the contrary. In 2008, world biofuel production reached 46 Mtoe, 65 Gl of ethanol (beyond the prediction in Fig. 25) and 16 Gl of biodiesel (ENERS, 2009).

The IEA Alternative Policy Scenario incorporates 1400 different policies and measures that aim at enhancing energy security and mitigating CO<sub>2</sub> emissions.

Measures in the transport sector would produce close to 60% of all oil savings in this scenario, more than two-thirds linked to more efficient new vehicles, the rest being related to the increased use of biofuels. With a faster assumed growth of 8.3% per year, biofuel production could rise to 6.15 EJ yr<sup>-1</sup> (147 Mtoe) by 2030. In the Reference or Alternative Scenarios, biofuels meet 4% or 7% of the world road-transport fuel in 2030, occupying 2.5% (34.5 Mha) or 3.8% (52.8 Mha) of world total agricultural land, respectively (Figs. 26, 27). If cellulosic ethanol were to be largely available by 2030, a larger share of biofuels (10%) could be possible with only a little extra land area (+ 5.7 Mha).



Fig. 27 Land requirements in Mha and percent of respective agricultural land areas for biofuels production in the different scenarios from the IEA. Drawn from data in IEA (2006)

Ethanol is expected to account for most of the biofuel increase worldwide as production costs are expected to fall faster than those of biodiesel and as it is likely to become a more attractive option for fuel suppliers in Europe. The global share of biodiesel nonetheless will grow in both scenarios because of production increase in the US and Brazil; it could reach up to 15% of total biofuel use in both countries. In both scenarios, the biggest increase in biofuel production and consumption occurs in Europe and the US may become a sizeable net importer of biofuels. Brazil remains the biggest ethanol exporter but other countries such as Malaysia or Indonesia could also become biofuel exporters. However, the assumed development of today's still limited international trade in biofuels will depend on whether trade barriers are removed or not (IEA, 2006). Biofuel trade liberalisation would lower the prices of blended fuel and enhance total biofuel demand imports. The shift from domestic production to import from abroad would be rather significant in most European countries for ethanol, from around 95% of domestic production to an average 50% share in 2020 across member states.

Trade amongst European member states would also decrease, from around 5% to 2%. For biodiesel, the impact should be much lower as import tariffs are much lower than for ethanol (Boeters et al., 2008).

If the IEA Alternative Policy Scenario was successfully implemented, energy-related CO<sub>2</sub> emissions could be cut by 1.7 Gt or 5% in 2015, and 6.3 Gt or 16% in 2030 relative to the Reference Scenario; 12% of these savings would come from renewables (including biofuels), 10% from nuclear power and the remaining from more efficient production and use of energy. Despite these savings, global CO<sub>2</sub> emissions would still be 8 Gt higher in 2030 than they are today. Going beyond the Alternative Policy Scenario to keep emissions at current levels, i.e. saving these 8 Gt, would mean increasing electricity efficiency by 50% over the alternative scenario and to implement new technologies such as CO<sub>2</sub> Capture and Storage. In this last scenario, 1Gt more would be saved in the transport sector thanks to more efficient and cleaner vehicles, notably using twice as much biofuel as in the Alternative Policy Scenario (IEA, 2006).

# 6.5 Impact of Biofuels on Agricultural Commodity Prices

As a background assumption in most potential assessments, energy crops should not compete with food and feed crops for land. In practice, land areas that are allocated in the supply-driven studies to food crops because of their high yield potential could also be allocated to energy crops by farmers depending on the markets. In a sensitivity analysis of the impact of land allocation on the bioenergy potential, 30% to 51% of the most productive land areas previously allocated to food and feed crops would then be allocated to energy crops, leading to an overall decrease in bioenergy potential in most cases due to the need for more agricultural land to produce enough food and feed on less productive areas. On the contrary, a global geographical optimisation of land-use patterns by allocating the most productive land areas to food crops would result in an increase in bioenergy potential in all scenarios (Smeets et al., 2007).

In 2006, global cereal stocks, especially wheat, were at their lowest levels since the early 1980s. As a result of reduced plantings and adverse weather in some major producing countries, world cereal production decreased by 2.4% between 2005 and 2006, coinciding with further expansion of the demand. Commodity prices hence rose steeply, further pushed up by speculative transactions adding to increased commodity-price volatility (Von Braun, 2007). Despite a more favourable global production outlook for the coming years, prices are unlikely to return to the low levels of previous years due to a host of reasons, notably the escalation of cost of inputs and the need to replenish stocks (FAO, 2008a). Biofuel production has contributed to the changed world food equation. While cereal use for food and feed has increased by 4% and 7%, respectively, since 2000, the use of cereals for industrial purposes, such as biofuel production, has increased by more than 25%; in the United States alone, the use of maize for ethanol production increased 150% between 2000 and 2006 (Earth Policy Institute and FAO in Von Braun, 2007). As a consequence, the prices of commodities used in biofuel production are becoming increasingly linked to energy prices. The coefficient of variation of oilseed price in the past five years was  $0.2^{64}$ , compared with typical

coefficients in the range of 0.08–0.12 in the past two decades; that of maize has increased from 0.09 to 0.22 in the past decade (Von Braun, 2007).

Since feedstock represents the principal share of total costs, the biofuel sector will both contribute to feedstock price changes and be affected by them. Hence, the impact of biofuel expansion on projected food prices is not yet well established (Von Braun, 2007). Furthermore, due to competitive land uses, biofuels may also impact prices of other food commodities than their own feedstock. Higher prices for maize, for instance, caused food consumers to shift to rice and wheat, while it was more profitable for producers to preferentially grow maize over these same crops. Prices of rice and wheat therefore increased. Comparing the expected grain price increase between 2000 and 2007 with and without the biofuel growth that occurred at that time, IFPRI estimated that biofuels were responsible for 30% of the increase in weighted average grain prices (Rosegrant, 2008).

Using the OECD/FAO Aglink/Cosimo/Sugar model (OECD, 2006), impacts on commodity prices until 2014 were modelled with three biofuel development scenarios compared with a baseline scenario, i.e. status quo agricultural policies, normal weather patterns, and biofuel growth in the US and Brazil only: the constant biofuel scenario with no growth in biofuel production, the policy-target scenario with projected biofuel growth in line with officially stated goals (in 2004), and a third scenario considering the policy-target scenario and higher oil prices (Fig. 28).

The biofuel growth in the baseline scenario has a relatively small impact on coarse grain, wheat and vegetable oil prices, increases of 2.5%, 1% and 1.5%, respectively, compared with the constant biofuel scenario. Impact on livestock markets prices is also limited despite slight price decreases due to the production of co-products such as oilseed meal for feed. The impact on sugar price is much larger, a 37% price increase up to 60% in the policy-target scenario.

The policy-target scenario especially requires enhanced biofuel production in the EU and Canada added to increased production in the US and Brazil. As a consequence, substantial growth in feedstock needs implies significant trade pattern changes. EU imports of vegetable oils would increase three-fold, while wheat exports would fall by 41%, and Canadian wheat and coarse grain exports would also decrease by 34% in 2014. Overall, world prices for most

<sup>&</sup>lt;sup>64</sup> i.e. 20% around the annual mean price.



**Fig. 28** World crop market prices under alternative scenario assumptions (OECD, 2006). Note: "Crude oil US\$60\*" denotes a scenario assuming higher

crude oil prices, but unchanged petrol-based fuel prices (and hence unchanged biofuel prices) relative to the policy-target scenario

commodities increase substantially compared with the constant biofuel scenario, by 4% for wheat, and 15% for vegetable oils, also inducing an increase in butter price of 3% as a substitute for oil (OECD, 2006). Despite a more significant decrease in oilseed meal price (-6%), meat market prices increase with and without growth in biofuel production. With crude oil price at a sustained level of US\$ 60 per barrel from 2005 to 2014 instead of US\$ 45 to US\$ 35 in the other scenarios, commodity prices first increase due to the high production costs. Biofuel costs also increase, but compared with the policy-target scenario, at the same time oil price provides a further incentive for biofuels, whose share would increase to 6% in 2014, compared with 5.5% in the policy-target scenario. The increase in biofuel production would lead to a further increase in commodity prices (OECD, 2006). In comparison, IFPRI modelled that in a 2007-baseline biofuel production, maize and oil prices would be higher by 6% and wheat and sugar by 4% in 2014 compared with a constant biofuel scenario (Rosegrant, 2008).

By 2020, taking current biofuel investment plans (2007) into account, international prices could increase by 26% for maize, by 18% for oilseeds, by 11% for

sugar, and by 8% for wheat (Von Braun, 2007). While the large response of sugar price to biofuel production is suspected to be inaccurate (note 35 in (OECD, 2006)), the lower response in the IMPACT-WATER model (IFPRI) is also surprising since sugar price is highly related to ethanol and energy prices, unless sugar price increase is limited in IMPACT-WATER following the WTO sugar reform or other factors. In Boeters et al. (2008), the impacts on food prices of a 10%-biofuel target turn out to be negligible on the European Union scale compared with the policy baseline, i.e. the current economic welfare already impacted by distortionary taxes.

Market price projections and comparisons remain superficial, as many factors are still highly uncertain, such as food and feed demand, speculations on commodity prices, and biofuel international trade, including the future role of developing countries, and the development of 2nd generation biofuels. The latter are expected to be produced mainly from residues or dedicated energy crops that will not lead to an additional demand for food commodities as feedstock. The use of marginal lands could lower the pressure on agricultural lands and therefore the land prices. However, this potential remains limited due to low productivity, high costs and potential impacts on landscape.

Where dedicated crops would take over food crops for the use of land, the impact on food prices would also depend on local policies. For instance, within the framework of the US 1996 Farm Bill, a scenario of switchgrass production substituting 4-9.5 Mha allocated to food crops between 1996 and 2000 could have led to a price increase of from 4-14% of maize, sorghum, wheat, soybean, cotton and rice. However, through higher farm income and reduced loan deficiency payments, switchgrass promotion could have led to significant savings for the treasury (Ugarte and Walsh, 2002), the bottom-line cost being passed on to the consumer prices. As an example, according to OECD estimations, the CAP cost to ordinary citizens is around 100 billion  $\in$  each year (US\$ 125 billion), half from taxpayers and half from consumers owing to higher food prices. This is an average cost to an EU family (4 people) of around  $950 \in$  a year (US\$ 1190), with only around  $20 \in (US\$ 25)$  of this spent as EU money on targeted environmental programmes. The CAP has been estimated to be equivalent to a value added tax on food of around 15% and removing market price support would bring a one-off reduction in inflation of 0.9% (OECD, 2005).

## 7 Conclusion

Although bioenergy and biofuels in particular have recently been high on the policy agenda and subject to a lot of discussion, they still only contribute a marginal share in the global energy supply. Some key points may help to figure out how bioenergy can play a bigger part in the years to come.

 When reviewing biomass potential assessments, a rather modest assumption would be that the share of bioenergy in the total energy consumption can be multiplied by at least a factor of 2.5, but such deployment scenarios are extremely hard to predict since technology evolution is non-linear<sup>65</sup>. This means breakthroughs may be expected along the way, notwithstanding the driving role of policies that are also constantly evolving. However, it remains certain that the contribution of biofuels will be limited because of the scarcity of available land, unless advances in technology make tremendously higher yields possible, in biomass production and conversion, with significantly lower costs. For second generation biofuels, a ten-fold increase in the plant productivity is still needed to reach commercial potential. A global 10% share of transportation fuels, excluding international aero-traffic, may be reached by 2020-2030, provided that biofuel chains are optimised in terms of both environmental and economic performance, and combined with changes in the automotive sector towards: lighter cars, hybrids, flex-fuel and city-vehicles. Taking the 13.5% contribution of transport to global CO<sub>2eq</sub> emissions, total emissions in 2000 of 41.75 GtCO<sub>2eq</sub> (Baumert et al., 2005), and a range of greenhouse gas savings of 20-60% (FAO, 2008a), 10% biofuel share would result in a reduction of 113-340 MtCO<sub>2eq</sub> per year. If second generation biofuels become available by 2030 and are combined with hybrid technology, biofuels could save an additional 1 GtCO<sub>2eq</sub> per year (IEA, 2006). This may represent a small contribution, but still a necessary one given the current trends and the scope of the GHG reduction targets (Fig. 29).

- The sooner second generation biofuels are commercially available, the more likely the 10% target will be met. Although they out-perform first generation biofuels, the latter will develop further and serve as a springboard for the second generation biofuels to be quickly introduced within well-established biofuel chains. Therefore, attention must be paid to the diffusion of best available practices and to the attainment of sustainability standards of first generation biofuel chains.
- From a CO<sub>2</sub> perspective, the best mitigation pathway is to prevent emissions. A combusted litre of biofuel will never perform better for the environment that a non-combusted litre of fossil fuel. However, a significant part of global greenhouse gas emissions cannot be avoided and for those only CO<sub>2</sub> or C sequestration can have a mitigative effect. From a sequestration point of view, the interest of biofuels is to concentrate non-point sources from transport into facilities where they can be captured, as is the case with CO<sub>2</sub> from sugar fermentation, for instance. It can be expected that with the need for

<sup>&</sup>lt;sup>65</sup> The Schumpeterian vision of technology advances that evolve by plateaus punctuated by radical breakthroughs.



**Fig. 29** Necessary reduction in greenhouse gas emissions (including from deforestation and other land use) to stay within the 2°C global warming target from today's perspective (Fischedick et al., 2007)

reducing industrial emissions, biofuel conversion facilities will keep improving in efficiency and gas savings or storage. Agricultural practices may also contribute to enhancing soil carbon sequestration; but while the residence time of stored soil carbon is not permanent and not easily controllable, the turnover of fixed atmospheric CO<sub>2</sub> by biomass and released through combustion can be quantified and in principle repeated indefinitely. Here, the intrinsic interest of biomass is that photosynthesis will be enhanced as atmospheric CO<sub>2</sub> concentration increases, provided that no other production factors are limiting, creating a negative feed-back loop. Finally, the overall interest of biofuels in terms of CO<sub>2eq</sub> savings relies on the overall performances of agro-ecosystems, which in most cases and for all agricultural production can be largely improved.

• Soil organic carbon lost through deforestation negates the benefit of biofuel in terms of CO<sub>2</sub> savings. However, merely shifting the burden of deforestation and biodiversity losses onto biofuels will not stop land clearing for agricultural purposes. On the contrary, funds for biofuel development programmes could provide leverage to implement sustainability criteria for agricultural production worldwide, and to enhance the productivity of traditional slash and burn cultivation, thereby preserving the forest. Moreover, policies are also needed, notably in tropical regions, to empower local populations to prevent illegal logging and effectively urge forest preservation and investment

in productive and environmental-friendly agroecosystems. In Brazil, for instance, it is cheaper to clear new land areas for the international beef and soya bean markets than to invest in already deforested regions<sup>66</sup>.

All bioenergy chains are not suitable for all locations. Bioenergy chains can bring benefits to the society in terms of fossil energy savings, as well as other positive environmental impacts, but only if the best appropriated bioenergy chain mix is chosen in accordance with local conditions, notably the biomass production systems and the types of primary energy inputs for conversion. Better knowledge of life-cycle greenhouse gas emissions from all energy uses of biomass, and strong sustainability criteria for biomass production, also addressing trade-off effects due to indirect land-use change, are still needed to fully assess the benefits and limitations of biomass use (EEA, 2008). Given the wide range of candidate biofuel chains, options will have to be identified and taken that minimise adverse environmental impacts while harnessing most of the advantages of biofuels. The resulting bioenergy mix will be systematically better for the environment than the business-as-usual scenario.

<sup>&</sup>lt;sup>66</sup> Dr. Emily Boyd, 25/11/2005 http://www.scidev.net/en/ opinions/emissions-trading-cannot-solve-amazon-deforestatio. html.

In countries where land area is the main limiting factor, for instance, the priority should be given to biofuels from waste oils, animal grease, residues and municipal waste. The competition for the use of biomass for heat and power or biofuel should also be limited as far as possible by giving priority to the bioenergy chain that makes it possible to save the most greenhouse gases compared with the local substituted energy source.

- In the case of sustainable bioenergy chains, positive externalities ought to be given economic values so that bioenergy could be more competitive. Overall benefits from bioenergy chains have to be considered, taking into account the value of all co-products. Considering the competition for natural resources, the principle of "zero waste" within integrated biorefinery appears to be the best economic and environmental choice. The use of contaminated or degraded lands for bioenergy purposes is also essential. Economic incentives should thus aim at fostering these priorities.
- The "success story" of ethanol in Brazil suggests that further growth in biofuel production can be expected through intensive breeding programmes to foster the development of second generation biofuel feedstock. Indeed, the high productivity of sugar cane has benefited from decades of research and commercial cultivation. Nowadays, cane growers in Brazil use more than 500 commercial cane varieties that are resistant to many of the crop diseases found in the country. Between 1975 and 2000, in the São Paulo state, the sugar cane yield per hectare increased by 33%, ethanol yield from sugar by 14%, and fermentation productivity by 130% (Kojima and Johnson, 2005). Another key element is the flexibility of the production unit, that has to be found as a balance between complete integration to reach maximum efficiency, minimum losses and economies of scale, while this optimum-oriented specialisation will not prevent the unit from being flexible enough to adapt the processes to various feedstocks and end-products. Most distilleries in Brazil are part of sugar mill/distillery complexes capable of switching between 60%-40% and 40%-60% sugar-ethanol, which makes it possible to take advantage of fluctuations in the relative prices of sugar and ethanol. In France for instance, mill/distillery complexes are optimised to produce 66% sugar and 33% ethanol on a year-run basis,

with very little flexibility<sup>67</sup>. The success of the mill/distillery complexes also relies on good integration within both the nationwide ethanol supply system and the electricity grid to sell the co-produced electricity, once their auto-consumption is satisfied (Kojima and Johnson, 2005).

Bioenergy can contribute to tackling part of the energy dependency and the depletion of nonrenewable resources, but they alone will not suffice. Their development would also be vain if at the same time energy efficiency was not drastically improved and energy consumption behaviours did not change radically. As the transport sector is the main growing source of energy consumption and greenhouse gas emissions, biofuels play a critical role as long as the bulk of vehicles is not electrified. Despite improvements in the energy eff iciency of various transport modes and the introduction of non-fossil fuels, increased transport demand, especially increased car usage and a reduced number of passengers per car, is outweighing these benefits. Present knowledge indicates that it will not be possible to achieve ambitious targets comparable with the Bali roadmap without limiting transport demand (EEA, 2008). In a recent survey, responses from 25767 EU citizens indicated that 54% would be willing to pay more for using less-polluting transport. The best ways to reduce transport CO<sub>2</sub> emissions and to promote biofuels would be through an interdiction to sell "polluting vehicles" that do not achieve stateof-the-art emission standards, and tax incentives to foster both fuel-efficient vehicles and biofuels (Fig. 30, Eurobarometer, 2007). However, evidence also suggests that only a minority of individuals actually take action to reduce private transport energy consumption and fewer may intend to take action in the future. Analysing "Special Eurobarometer" surveys from 1984, 1993 and 2002, it appears that action concerning reducing car fuel use has not increased between 1993 and 2002, despite an increase in real fuel prices, and intentions to take action to reduce energy use were generally lower in 2002 than in 1993 or 1984. According to the 2002 survey, more than half of all respondents (55%) reported having taken no action on energy

<sup>&</sup>lt;sup>67</sup> Personal communication by Dane Colbert, Director of Ethanol Union, 30/10/2008.



**Fig. 30** (a) The best way to reverse the rise of  $CO_2$  emissions from road transport in response to question Q5: road transport generates about one fifth of the European Union's harmful emissions. Between 1990 and 2004,  $CO_2$  emissions from road transport rose by 26%. Which is the best way to reverse this trend?

efficiency in any of the four broad transport energy efficiency measures examined (reducing travel, cutting fuel use, buying a more efficient vehicle and using public transport) while almost two-thirds of all respondents (64%) report that they do not intend to take further action in any of these areas of energy efficiency (Stead, 2007). Regarding the fact that awareness of transport pollution issues is, moreover, likely to be lower on a global scale than in Europe, radical measures may be necessary in order to radically change transport consumer behaviours. To address transport demand, measures and policy instruments must hence also go beyond the transport sector itself and be introduced into sectors of the economy such as households, industry and service, within which the demand for transport actually originates (EEA, 2008).

• Harmonisation at an international level is crucial in order to ensure the overall complementarity of bioenergy chains, to provide a coherent framework for the markets and to control the sustainability of the systems. In particular, the framework for international biofuel trade is complex: trade barriers should be removed so that no artificial competitiveness would hamper the development of biofuels, but on the other hand, control of sustainability is necessary on a global scale and instruments should be put in place to ensure it. Voluntary schemes for the certification of sustainable biomass already exist, e.g. the Forest Stewardship Council (FSC) b Tax incentives to make bio fuel cheaper Compulsory standards for manufacturers to produce cars that use bio fuel Crop subsidies for bio fuel production Higher taxes for polluting vehicles using traditional fossil fuel Other measures DK/NA 7

(b) The best way to encourage the use of biofuels in response to question Q8: biofuels are renewable fuels that can reduce fossil oil dependence of vehicles. Which is in your opinion the best measure to encourage the use of bio fuels? %, Base: all respondents DK: Don't Know/NA: No Answer (Eurobarometer, 2007)

for forest product or the Roundtable Sustainable Palmoil (RSPO), and are currently being discussed as implementation options for bioenergy sustainability standards. Hence most of the key elements for such standards are available as well as experiences from existing voluntary schemes. Although legally binding standards are superior, pragmatically, voluntary schemes might provide a well-needed start ("entry option") (Fritsche et al., 2006). Finally, cross-sector integration of agriculture, energy and transport policies is also mandatory to make biofuel incentives coherent and to send a clear message to the population.

Concerns about GMOs and rising food prices are justified. However, these are not exclusive to biofuels, and should be addressed in a wider perspective. Not mentioning the debate on GMO, competition for land uses between food and non-food crops should be minimised as much as possible. Where competition for land uses is critical, market distortion should not, as in the case of the cotton market, spoil opportunities given to developing countries. It is important that governmental support for biofuels as an infant industry remains temporary, or else the policy will result in inefficient allocation of resources in the long run, once costs decline as output expands and production experience is acquired. The extent to which biofuel programmes can contribute to rural development is dependent on the industry characteristics and, ultimately, whether it is able to become financially viable without direct government support. However, if public funds are needed to support the industry, the question to be addressed in the first place is whether government resources will be diverted from other programmes and what would be the comparative impacts on rural development and the environment (Kojima and Johnson, 2005). From an agronomical point of view, promising options in developing countries are in particular those that introduce energy crops within agro-forestry systems. However, the need for an intensification of the production, and also the development of perennial plantations, may remain largely hampered as long as the lack of land property rights does not make it possible to empower the farmers. Biofuel programmes need to be integrated within a broader context of investment in rural infrastructure and human capital formation. Indeed, strengthening property rights, removing both international and domestic trade-barriers, access to education, water, electricity and networks, and developing transport infrastructure have proven better drivers of rural development than direct aid in providing farming equipment or inputs, for instance (Kojima and Johnson, 2005).

## References

- ADEME (2006) Programme national Bois-Énergie 2000-2006 (National Wood Energy Programme), Rapport d'activités 2000-2005 (Activities Report), Collection : Données et références, ©ADEME Editions, 5853, ISBN 978-2-86817-844-8, Paris, 114 p.
- ADEME/DIREM (2002) Bilans énergétiques et gaz à effet de serre des filières de production de biocarburants, Rapports techniques, Version définitive novembre 2002, Ecobilan, Pricewaterhouse Coopers, 132 p.
- Agard J., Alcamo J., Ash N., Arthurton R., Barker S., Barr J., Baste I., Chambers W.B., Dent D., Fazel A., Gitay H., Huber M., Jäger J., Kuylenstiema J.C.I., King P.N., Kok M.T.J., Levy M.A., Mafuta G., Martino D., Panwar T.S., Rast W., Rothman D.S., Varughese G.C., Zommers Z. (2007) Global Environment Outlook GEO4 Environment for Development, UNEP, ISBN: 978-92-807-2836-1, 572 p.
- Anton A., Castells F., Montero J.I. (2007) Land use indicators in life cycle assessment. Case study: The environmental impact of Mediterranean greenhouses, J. Cleaner Prod. 1, 432–438.
- Antoni V., Arrouays D. (2007) Le stock de carbone dans les sols agricoles, IFEN 121, 4 p.
- Arrouays D., Balesdent J., Germon J.C., Jayet P.A., Soussana J.F., Stengel P. (2002) Mitigation of the greenhouse effect Increasing carbon stocks in French agricultural soils?

Scientific Assessment Unit for Expertise, Synthesis of an Assessment Report by the French Institute for Agriculture Research (INRA) on request of the French Ministry for Ecology and Sustainable Development, October 2002, 33 p.

- Arrouays D., Deslais W., Badeau V. (2001) The carbon content of topsoil and its geographical distribution in France, Soil Use Manage. 17, 7–11.
- ASPO (2008) Newsletter February 2008, online: http://www. peakoil.net.
- Bai Z.G., Dent D.L., Olsson L., Schaepman M.E. (2007) Global Assessment of Land Degradation and Improvement, FAO LADA working paper, Food and Agriculture Organization of the United Nations, Rome.
- Bai Z.G., Dent D.L., Olsson L., Schaepman M.E. (2008) Global Assessment of Land Degradation and Improvement. 1 Identification by remote sensing, Report 2008/01(GLADA Report 5), ISRIC – World Soil Information, Wageningen, 70 p.
- Balat M, Balat H., Öz C. (2008) Progress in bioethanol processing, Progress in Energy Combustion Science, DOI: 10.1016/j.pecs.2007.11.001.
- Balesdent J., Chenu C., Balabane M. (2000) Relationship of soil organic matter dynamics to physical protection and tillage, Soil Till. Res. 53, 215–230.
- Ball B.C., Crichton I., Horgan G.W. (2008) Dynamics of upward and downward N<sub>2</sub>O and CO<sub>2</sub> fluxes in ploughed or no-tilled soils in relation to water-filled pore space, compaction and crop presence, Soil Till. Res. 101, 20–30.
- Bange H.W. (2006) New directions: The importance of oceanic nitrous oxide emissions, Atmos. Environ. 40, 198–199.
- Barbier C., Baron R., Colombier M., Boemare C. (2004) Climate change policies; Analysis of sectoral changes in Europe, ID-DRI/CIRED with the support of ADEME, Paris, 37 p.
- Bare J.C., Hofstetter P., Pennington D., Udo de Haes H.A. (2000) Life Cycle Impact Assessment Workshop Summary: Midpoints vs Endpoints – The Sacrifices and Benefits, Int. J. LCA 5, 319–326.
- Baserga U. (2000) Fermentation de déchets organiques dans les installations agricoles de biogaz, Station fédérale de recherches en économie et technologie agricoles (FAT) CH-8356 Tänikon, FAT Rapports n° 546, 12 p.
- Basset-Mens C., Anibar A., Durand P., van der Werf H. M.G. (2006a) Spatialised fate factors for nitrate in catchments: Modelling approach and implication for LCA results, Sci. Total Environ. 367, 367–382.
- Basset-Mens C., Van der Werf H.M.G., Durnad P., Leterme P. (2006b) Implications of Uncertainty and Variability in the Life Cycle Assessment of Pig Production Systems, Int. J. LCA 11, 298–304.
- Baumert K.A., Herzog T., Pershing J. (2005) Navigating the Numbers; Greenhouse Gas Data and International Climate Policy. © World Resources Institute, ISBN: 1-56973-599-9, USA, 132 p.
- Beheydt D., Boeckx P., Sleutel S., Li C., van Cleemput O. (2007) Validation of DNDC for 22 long-term N<sub>2</sub>O field emission measurements, Atmos. Environ. 41, 6196–6211.
- Berndes G., Hoogwijk M., Van den Broek R. (2003) The contribution of biomass in the future global energy supply: a review of 17 studies, Biomass Bioenerg. 25, 1–28.
- Bioenergy Business (2007) Markets and finance for biomass and biofuels, Volume 1, Issue 3, 23 p.

- BIOFRAC (2006) Biofuels in the European Union: a vision for 2030 and beyond, Final draft report of the BIOFRAC, European Biofuels Research Advisory Council, 30 p.
- Björklund A. (2002) Survey of approaches to improve reliability in LCA, Int. J. LCA 7, 64–72.
- Blottnitz von H., Curran M.A. (2007) A review of assessments conducted on bio-ethanol as a transportation fuel from a net energy, greenhouse gas, and environmental life cycle perspective, J. Cleaner Prod. 15, 607–619.
- Boer de I.J.M. (2003) Environmental impact assessment of conventional and organic milk production, Livest. Prod. Sci. 80, 69–77.
- Boeters S., Veenendaal P., van Leeuwen N., Rojas-Romagoza H. (2008) The potential for biofuels alongside the EU-ETS, Paper for presentation at the Eleventh Annual GTAP Conference 'Future of Global Economy', Helsinki, June 12–14 2008, 39 p.
- Boizard H., Richard G., Roger-Estrade J., Dürr C., Boiffin J. (2002) Cumulative effects of cropping systems on the structure of the tilled layer in northern France, Soil Till. Res. 64, 149–164.
- Bouwman A.F. (1994) Method to estimate direct nitrous oxide emissions from agricultural soils, Report 773004004, National Institute of Public Health and Environmental Protection, Bilthoven, the Netherlands, 28 p.
- BRDI (2006) Vision: for bioenergy and biobased products in the United States, Bioeconomy for a sustainable future, Biomass Research and Development Initiative, 27 p.
- Brentrup F., Küsters J., Kuhlmann H., Lammel J. (2004) Environmental impact assessment of agricultural production systems using the life cycle assessment methodology. I. Theoretical concept of a LCA method tailored to crop production, Eur. J. Agron. 20, 247–264.
- Cardona C.A., Sanchez O.J. (2007) Fuel ethanol production: Process design trends and integration opportunities, Bioresource Technol. 98, 2415–2457.
- CEA (2004) L'hydrogène, les nouvelles technologies de l'énergie, Commissariat de l'Énergie Atomique, Clefs n° 50/51, 156 p.
- CEC (1997) Commission of the European Communities, Energy for the Future: Renewable Sources of Energy, White Paper for a Community Strategy and Action Plan, COM(97) 599 final (26/11/1997) Brussels, 55 p.
- CEC (2005) Commission of the European Communities, Biomass Action Plan, COM(2005) 628 final, {SEC(2005) 1573} Brussels, 47 p.
- CEC (2006a) Commission of the European Communities, EU Strategy for Biofuels, COM(2006) 34 final, {SEC(2006) 142} Brussels, 30 p.
- CEC (2006b) Commission of the European Communities, a European Strategy for Sustainable, Competitive and Secure Energy, Green Paper COM(2006) 105 final, {SEC(2006) 317} Brussels, 20 p.
- CEC (2008) Commission of the European Communities, Proposal for a Directive of the European Parliament and of the Council on the promotion of the use of energy from renewable sources; (presented by the Commission) {COM(2008) 19 final} 2008/2016 (COD), Brussels, 61 p.
- Chapple C., Ladisch M., Meilan R. (2007) Loosening lignin's grip on biofuel production, Nat. Biotechnol. 25, 746–748.

- Chapuis-Lardy L., Wrage N., Metay A., Chottes J.L., Bernoux M. (2007) Soils, a sink for N<sub>2</sub>O? A review, Global Change Biol. 13, 1–17.
- Chatskikh D., Olesen J. (2007) Soil tillage enhanced CO<sub>2</sub> and N<sub>2</sub>O emissions from loamy sand soil under spring barley, Soil Till. Res. 97, 5–18.
- Chenu C., Le Bissonnais Y., Arrouays D. (2000) Organic Matter Influence on Clay Wettability and Soil Aggregate Stability, Soil Sci. Soc. Am. J. 64, 1479–1486.
- Clem A. (1985) Commodity price volatility: trends during 1975–1984, Monthly Labor Review, 5 p.
- Clifton-Brown J.C., Breuer J., Jones M.B. (2007) Carbon mitigation by the energy crop, *Miscanthus*, Global Change Biol. 13, 2296–2307.
- Conen F., Dobbie K.E., Smith K.A., (2000) Predicting N<sub>2</sub>O emissions from agricultural land through related soil parameters, Global Change Biol. 6, 417–426.
- Conrad R. (1990) Flux of NO<sub>x</sub> between soil and atmosphere: importance and soil microbial metabolism, in: Revsbech N.P., Sörensen J. (Eds.), Denitrification in soil and sediment, Plenum Press, New York, 1990.
- Conrad R. (1996) Soil microorganisms as controllers of atmospheric trace gases (H<sub>2</sub>, CO, CH<sub>4</sub>, OCS, N<sub>2</sub>O, and NO), Microbiol. Rev. 60, 609–640.
- Cowie A.L., Smith P., Johnson D. (2006) Does soil carbon loss in biomass production systems negate the greenhouse benefits of bioenergy? Mitigation and Adaptation Strategies for Global Change 11, 979–1002.
- Croezen H., Kampman B., Van de Vreede G., Sevenster M. (2007) ETBE and ethanol: a comparison of  $CO_2$  savings, Delft, CE, 63 p.
- Crutzen P.J., Mosier A.R., Smith K.A., Winiwarter W. (2008)  $N_2O$  release from agro-biofuel production negates global warming reduction by replacing fossil fuels, Atmos. Chem. Phys. 8, 389–395.
- Del Grosso S.J., Parton W.J., Mosier A.R., Hartman M.D., Brenner J., Ojima D.S., Schimel D.S. (2001) Simulated interaction of carbon dynamics and nitrogen trace gas fluxes using the DAYCENT model, in: Schaffer M., Ma L., Hansen S. (Eds.), Modeling Carbon and Nitrogen Dynamics for Soil Management, CRC Press, Boca Raton, Florida, pp. 303–332.
- Delucchi M.A. (2004) Conceptual and methodological issues in life cycle analyses of transportation fuels, Does Research and Research Scientist Institute of transportation Studies University of California, prepared for the U.S. Environmental Protection Agency Office of Transportation and Air Quality, p. 22.
- Demirbas A. (2007) Progress and recent trends in biofuels, Prog. Energ. Combust. 33, 1–18.
- Denman K.L., Brasseur G., Chidthaisong A., Ciais P., Cox P.M., Dickinson R.E., Hauglustaine D., Heinze C., Holland E., Jacob D., Lohmann U., Ramachandran S., da Silva Dias P.L., Wofsy S.C., Zhang X. (2007) Couplings Between Changes in the Climate System and Biogeochemistry, Climate Change 2007: The Physical Science Basis, Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Solomon S., Qin D., Manning M., Chen Z., Marquis M., Averyt K.B., Tignor M., Miller H.L. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

- Dornburg V., van Dam J., Faaij A. (2007) Estimating GHG emission mitigation supply curves of large-scale biomass use on a country level, Biomass Bioenerg. 31, 46–65.
- Duxbury J.M., Mosier A.R. (1993) Status and issues concerning agricultural emissions of greenhouse gases, in: Agricultural Dimension of Global Climate Change, Chap. 12, Sainte Lucie Press, pp. 229–258.
- EEA (2006a) Greenhouse gas emission trends and projections in Europe 2006, Environmental Energy Agency, Report 9, ISBN 92–9167–885-6 © EEA, Copenhagen, 63 p.
- EEA (2006b) How much bioenergy can Europe produce without harming the environment? Environmental Energy Agency, Report 7, ISBN 92–9167–849-X © EEA, Copenhagen, 72 p.
- EEA (2008) Climate for a transport change, TERM 2007: indicators tracking transport and environment in the European Union, EEA Report No. 1, ISSN 1725-9177, DOI 10.2800/3320 © EEA, Copenhagen, 56 p.
- ENERS (2009) Biofuels Platform, Online: http://www.eners.ch.
- Engel Th., Priesack E. (1993) Expert-N, a building block system of nitrogen models as a resource for advice, research, water management and policy, Integral Soil and Sediment Research: a Basis for Proper Protection, in: Eijsackers H.J.P., Hamers T (Eds.), Kluwer, Dordrecht, pp. 503–507.
- EU (2001) Directive 2001/77/EC of the European Parliament and of the Council, 27 September 2001, on the promotion of electricity produced from renewable energy sources in the internal electricity market, Official Journal of the European Union 27.10.2001, L 283/33, 8 p.
- EU (2003) Directive 2003/30/EC of the European Parliament and of the Council, 8 May 2003, on the promotion of the use of biofuels or other renewable fuels for transport, Official Journal of the European Union 17.5.2003, L 123/42, 5 p.
- EU (2007a) Brussels European Council 8/9 March 2007, Presidency conclusions.
- EU DG-TREN (2005) European Commission, Directorate-General for Energy and Transport, 20% energy savings by 2020, Green Paper on energy efficiency, Memo, 7 p.
- EU DG-TREN (2006a) European Commission, Directorate-General for Energy and Transport Fact sheet: EU-25, Energy Fiches, TREN C1, Online: http://ec.europa.eu/energy/ green-paper-energy/doc/2006\_03\_08\_gp\_factsheet\_en.pdf.
- EU DG-TREN (2006b) European Commission, Directorate-General for Energy and Transport, Fuelling our future, Green Paper for a European strategy for sustainable, competitive and secure energy, Memo, 6 p.
- EU DG-TREN (2007) Energy and transport in figures: statistical pocket book. Online: http://ec.europa.eu/dgs/energy\_ transport/figures/pocketbook/2007\_en.htm.
- Eurobarometer (2007) Attitudes on issues related to EU Transport Policy: Analytical report, The Gallup Organization, European Commission, 82 p.
- EurObserv'ER (2006) Biofuels Barometer, in: Systèmes Solaires n° 173, pp. 57–66.
- EuroCare (2004) Outlooks on selected agriculture variables for the 2005 State of the Environment and the Outlook Report, EEA/RNC/03/016.
- European Commission (2007) Report on the hearing of the Biofuels European Technology Platform, Joint Research Center Directorate-General, Institute for Energy, Energy Systems Evaluation Unit. Petten, 5 p.

- Fagernäs L., Johansson A.N, Wilen C., Sipilä K., Mäkinen T., Satu H., Daugherty E., Den Uil H., Vehlow J., Kåberger T.S., Rogulska M. (2006) Bioenergy in Europe: Opportunities and Barriers, Bioenergy NoE, VTT Research Notes 2352, Espoo 2006, ISBN 951-38-6815-X, Valopaino Oy, Helsinki 2006, 122 p.
- FAO (2001) Global estimates of gaseous emissions of NH<sub>3</sub>, NO and N<sub>2</sub>O from agricultural land, ISBN 92-5-104698-1, International Fertilizer Industry Association, Food and Agriculture Organization of the United Nations, Rome.
- FAO (2002) World agriculture: toward 2015/2030, Summary report, ISBN 92-5-104761-8, Rome, 95 p.
- FAO (2003) FAO stat database on http://apps.fao.org/page/ collections, Food and Agriculture Organization of the United Nations: Rome.
- FAO (2007) Food Outlook, Global market analysis, GIEWS Global Information and Early Warning System on Food and Agriculture, 91 p.
- FAO (2008a) The state of food and agriculture. Biofuels: prospects, risks and opportunities, ISBN 978-92-5-105980-7, Rome, 138 p.
- FAO (2008b) Current world fertilizer trends and outlook to 2012, Rome, 34 p.
- Fargione J., Hill J., Tilman D., Polasky S., Hawthorne P. (2008) Land clearing and the biofuel carbon debt, Science 319, 1235–1238.
- Farquharson R., Baldock J. (2008) Concepts in modelling N<sub>2</sub>O emissions from land use, Plant Soil 309, 147–167.
- Farrell A.E., Plevin R.J., Turner B.T., Jones A.D., O'Hare M., Kammen D.M. (2006) Ethanol can contribute to energy and environmental goals, Science 311, 506–508.
- Firestone M.K., Davidson E.A. (1989) Microbiological basis of NO and  $N_2O$  production and consumption in soil, in: Exchange of trace gases between terrestrial ecosystems and the atmosphere, ISBN 0-471-92551-9, pp. 7–21.
- Fischedick M., Esken A., Luhmann H.-J., Schüwer D., Supersberger N. (2007) CO<sub>2</sub>-Capture and Geological Storage as a Climate Policy Option: technologies, Concepts, Perspectives, Wuppertal Spezial 35, 34 p.
- Fischer G., Schrattenholzer L. (2001) Global bioenergy potentials through 2050, Biomass Bioenerg. 20, 151–159.
- Fisher M.J., Rai I.M., Ayarza M.A., Lascano C.E., Sanz J.I., Thomas R.J., Vera R.E. (1994) Carbon storage by introduced deep-rooted grasses in the South American savannas, Nature 371, 236–238.
- FNR (2007) Biokraftstoffe, Pflanzen, Rohstoffe, Produkte, Stand: März 2008, Fachagentur Nachwachsende Rohstoffe e.v. (FNR) nachwachsende-rohstoffe.de, 22 p.
- FNR (2008) Biogas Basisdaten Deutschland. Stand: Oktober 2008, Fachagentur Nachwachsende Rohstoffe e.v. (FNR) nachwachsende-rohstoffe.de, 7 p.
- Forster P., Ramaswamy V., Artaxo P., Berntsen T., Betts R., Fahey D.W., Haywood J., Lean J., Lowe D.C., Myhre G., Nganga J., Prinn R., Raga G., Schulz M. Van Dorland R. (2007) Changes in Atmospheric Constituents and in Radiative Forcing, in: Solomon S., Qin D., Manning M., Chen Z., Marquis M., Averyt K.B., Tignor M., Miller H.L. (Eds.), Climate Change 2007: The Physical Science Basis, Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change,
Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

- France Betteraves/Passion Céréales (2007) La filière bioéthanol, États des lieux et perspectives, 36 p.
- Frank A.B., Berdahl J.D., Hanson J.D., Liebig M.A., Johnson H.A. (2004) Biomass and carbon partitioning in switchgrass, Crop Sci. 44, 1391–1396.
- Fritsche U.R., Hünecke K., Hermann A., Schulze F., Wiegmann K., Adolphe M. (2006) Sustainability Standards for Bioenergy, in: Lübbeke I. (Ed.), Published by WWF Germany, Frankfurt am Main and Öko-Institut e.V., Darmstadt, 80 p.
- Frolking S.E., Mosier A.R., Ojima D.S., Li C., Parton W.J., Potter C.S., Priesack E., Stenger R., Haberbosch C., Dörsch P., Flessa H., Smith K.A. (1998) Comparison of N<sub>2</sub>O emissions from soils at three temperate agricultural sites: simulation of year-round measurements by four models, Nutr. Cycl. Agroecosys. 52, 77–105.
- Fulton L., Howes T., Hardy J. (2004) Biofuels for transport, an international perspective, IEA study undertaken by the Office of Energy Efficiency, Technology and RandD, Paris, 210 p.
- Gabrielle B., Gagnaire N. (2008) Life-cycle assessment of straw use in bio-ethanol production: a case-study based on deterministic modelling, Biomass Bioenerg. 32, 431–441.
- Gabrielle B., Da-Silveira J., Houot S., Michelin J. (2005) Fieldscale modelling of carbon and nitrogen dynamics in soils amended with urban waste composts, Agr. Ecosys. Environ. 110, 289–299.
- Gabrielle B., Laville P., Hénault C., Nicoullaud B., Germon J.C. (2006) Simulation of nitrous oxide emissions from wheatcropped soils using CERES, Nutr. Cycl. Agroecosys. 74, 133–146.
- Galloway J.N., Aber J.D., Erisman J.W., Seitzinger S.P., Howarth R.W., Cowling E.B., Cosby B.J. (2003) The nitrogen cascade, BioScience, 53, 341–356.
- Galloway J.N., Townsend A.R., Erisman J.W., Bekunda M., Cai Z., Freney J.R., Martinelli L.A., Seitzinger S.P., Sutton M.A. (2008) Transformation of the Nitrogen Cycle: Recent Trends, Questions, and Potential Solutions, Science 320, 889–892.
- Garten C.T., Wullschleger S.D. (1999) Soil carbon inventories under a bioenergy crop (switchgrass): Measurement limitations, J. Environ. Qual. 28, 1359–1365.
- Germon J.C., Arrouays D., Deseau S., Félix I., Gabrielle B., Ganteil A., Gallienne J., Lellahi A., Lecorre N., Martin M., Mary B., Métay A., Nicolardot B., Quéré L., Reau R. (2007) Évaluation des impacts environnementaux des Techniques Culturales Sans Labour en France, Synthèse du groupe de travail sur le "bilan effet de serre ", ADEME-ARVALIS Institut du végétal-INRA-APCA-AREAS-ITB-CETION-IFVV, 400 p.
- Goedkoop M. (1995) NOH report 9523, The Eco-Indicator 95, Final Report, Pre' Consultants, Amersfoort.
- Goedkoop M., Spriensma R. (2000) The Eco-indicator 99: a damage oriented method for life cycle assessment, methodology report, second edition, Amersfoort, Netherlands: Pré Consultants, 132 p, online: http://www.pre.nl/ ecoindicator99/ei99-reports.htm.
- Goldberg S.D., Gebauer G. (2008) Drought turns a Central European Norway spruce forest soil from an N<sub>2</sub>O source to a tran-

sient N<sub>2</sub>O sink, Glob. Change Biol., DOI:10.1111/j.1365-2486.2008. 01752.x.

- Goodrum J.W., Geller D.P. (2005) Influence of fatty acid methyl esters frm hydoxylated vegetable oils on diesel fuel lubricity, Bioresource Technol. 96, 851–855.
- Graboski M.S. (2002) Fossil Energy Use in the Manufacture of Corn Ethanol, Colorado School of Mines, prepared for the National Corn Growers Association, 122 p.
- Graboski M.S, McCormick R.L. (1998) Combustion of fat and vegetable oil derived fuels in diesel engines, Prog. Energ. Combust. 24, 125–164.
- Grainger A. (1988) Estimating areas of degraded tropical lands requiring replenishment of forest cover, Int. Tree Crops J. 5, 31–61.
- Grainger A. (1991) Constraints on Increasing Tropical Forest Area to Combat Global Climate Change, in: Howlett D., Sargent C. (Eds.), Proceedings of the Technical Workshop to Explore Options for Global Forestry Management, Bangkok, 20–30 April, International Institute for Environment and Development, London, pp. 196–208.
- Granli T., Bøckman O.C. (1994) Nitrous oxide from Agriculture, Norwegian J. Agric. Sci. Suppl. 12, 1–128.
- Grant B., Smith W.N., Desjardins R., Lemke R., Li C.S. (2004) Estimated N<sub>2</sub>O and CO<sub>2</sub> emissions as influenced by agricultural practices in Canada, Climatic Change 65, 315–332.
- Gregorich E.G., Rochette P., Hopkins D.W., McKim U.F., St-Georges P. (2006) Tillage-induced environmental conditions in soil and substrate limitation determine biogenic gas production, Soil Biol. Biochem. 38, 2614–2628.
- Guinée J.B. (2002) Handbook on LCA; Operational Guide to the ISO Standards, ISBN 1-4020-0228-9, Kluwer Academic Publishers, Dordrecht, The Netherlands, 692 p.
- Guo L.B., Gifford R.M. (2002) Soil carbon stocks and land use change: a meta analysis, Global Change Biol. 8, 345–360.
- Hahn-Hägerdal B., Galbe M., Gorwa-Grauslund M.F., Lidén G., Zacchi G. (2006) Bio-ethanol – the fuel of tomorrow from the residues of today, Review Trends Biotechnol. 24, 549–556.
- Hansen E.M., Christensen B.T., Jensen L.S, Kristensen K. (2004) Carbon sequestration in soil beneath long-term *Miscanthus* plantations as determined by C-13 abundance, Biomass Bioenerg. 26, 97–105.
- Hansen J.E. (2006) Declaration of James E. Hansen, the Director of NASA Goddard Institute for Space Studies, The Earth Institute at Columbia University, on the 14th of August 2006 in front of the United States District Court for the District of Vermont Case Nos. 2:05-CV-302, and 2:05-CV-304 (Consolidated) retrieved (05/2007) from the web page: http://www. columbia.edu.
- Hansen J.E. (2007) Scientific reticence and sea level rise, Environ. Res. Lett. 2, 1–6. online: http://www.stacks.iop.org/ ERL/2/024002.
- Harijan K., Memon M., Uqaili M.A., Mirza U.K. (2009) Potential contribution of ethanol fuel to the transport sector of Pakistan, Renew. Sust. Energ. Rev. 13, 291–295.
- Heaton E.A. Clifton-Brown J. Voigt T.B. Jones M.B., Long S.P. (2004) *Miscanthus* for renewable energy generation: European union experience and projections for Illinois. Review, Mitigation and Adaptation Strategies for Global Change 9, 433–451.
- Hector R., Hughes S., Liang-Li X. (2008) Developing yeast strain for biomass-to-ethanol production, Ethanol

Producer Magazine, June 2008 Issue, online: http://www.ethanolproducer.com.

- Heijungs R., Goedkoop M., Struijs J., Effting S., Sevenster M., Huppes G. (2003) Towards a life cycle impact assessment method which comprises category indicators at the midpoint and the endpoint level, Report of the first project phase: Design of the new method VROM report, online: http://www. leidenuniv.nl/cml/ssp/publications/recipe\_phase1.pdf.
- Heijungs R., Guinée J.B., Huppes G., Lnakreijer R.M., Udo de Haes H.A., Wegener Sleeswijk A., Ansems A.M.M., Eggels P.G., van Duin R., de Goede H.P. (1992) Environmental Life Cycle Assessment of Products, Center of Environmental Science (CML), Leiden University, The Netherlands.
- Heinen M. (2006a) Simplified denitrification models: Overview and properties, Geoderma 133, 444–463.
- Heinen M. (2006b) Application of a widely used denitrification model to Dutch data sets, Geoderma 133, 464–473.
- Henao J., Baanante C. (2006) Agricultural Production and Soil Nutrient Mining in Africa – Implications for Resource Conservation and Policy Development, IFDC, Muscle Shoals, AL.
- Hénault C., Germon J.C. (2000) NEMIS, a predictive model of denitrification on the field scale, Eur. J. Soil Sci. 51, 257–270.
- Hénault C., Bizouard F., Laville P., Gabrielle B., Nicoullaud B., Germon J.C., Cellier P. (2005) Predicting in situ soil N<sub>2</sub>O emission using NOE algorithm and soil database, Global Change Biol. 11, 115–127.
- Higgins T. (2007) Study on relative CO<sub>2</sub> savings comparins ethanol and EtBE as a gasoline component, Final report, Hart Energy Consulting, 31 p.
- Himmel M.E., Ding S.-Y., Johnson D.K., Adney W.S., Nimlos M.R., Brady J.W., Foust T.D. (2007) Biomass Recalcitrance: Engineering Plants and Enzymes for Biofuels Production, Science 315, 804–807.
- Holmgren P. (2006) Global Land Use Area Change Matrix: Input to the Fourth Global Environmental Outlook (Geo- 4), Forest Resources Assessment, Working Paper 134, FAO, Rome, p. 9.
- Hoogwijk M., Faaij A., Eickhout B., de Vries B., Turkenburg W. (2005) Potential of biomass energy out to 2100, for four IPCC SRES land-use scenarios, Biomass Bioenerg. 29, 225–257.
- Hoogwijk M., Faaij A., Van den Broek R., Berndes G., Gielen D., Turkenburg W. (2003). Exploration of the ranges of the global potential of biomass for energy, Biomass Bioenerg. 25, 119–133.
- Houghton R.A. (2003) Revised estimates of the annual net flux of carbon to the atmosphere from changes in land use and land management 1850–2000, Tellus 55B, 378–390.
- Houghton R.A. (2008) Carbon Flux to the Atmosphere from Land-Use Changes: 1850–2005, in: TRENDS: A Compendium of Data on Global Change, Carbon Dioxide Information Analysis Center, Oak Ridge National Laboratory, U.S. Department of Energy, Oak Ridge, Tenn., U.S.A., online: http://cdiac.ornl.gov/trends/landuse/ houghton/houghton.html.
- Houghton R.A, Unruh J., Lefebvre P.A. (1991) Current land use in the tropics and its potential for sequestering carbon, in: Howlett D., Sargent C. (Eds.), Proc. Tech. Workshop to Explore Options for Global Forest Management, Bangkok, Thailand, London, IIED.

- Huijbregts M.A., Seppälä J. (2000) Towards region-specific, European fate Factors for airborne nitrogen compounds causing aquatic eutrophication, Int. J. LCA 5, 65–67.
- IEA (2005) World Energy Outlook 2005, Middle East and North Africa Insights, © OECD/IEA, IEA Publications (61 2005 26 1P1), ISBN 92-64-1094-98, Paris, 634 p.
- IEA (2006) World Energy Outlook 2006, © OECD/IEA, IEA Publications (61 2006 231P1), ISBN 92-64-10989-7-2006, Paris, 601 p.
- IEA (2007a) Key World Energy Statistics 2007, © OECD/IEA, 2007, IEA Publications, Paris, 82 p.
- IEA (2007b) Good practice guidelines, Bioenergy Project Development and Biomass Supply, © OECD/IEA, Paris, 62 p.
- IEA Bioenergy (2007) Potential Contribution of Bioenergy to the World's Future Energy Demand, IEA Bioenergy: ExCo, 12 p.
- IPCC (2006) IPCC Guidelines for National Greenhouse Gas Inventories, prepared by the National Greenhouse Gas Inventories Programme, in: Eggleston H.S., Buendia L., Miwa K., Ngara T., Tanabe K. (Eds.), IGES, Japan.
- IPCC (2007) Summary for Policymakers, Climate Change 2007: The Physical Science Basis, Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Solomon S., Qin D., Manning M., Chen Z., Marquis M., Averyt K.B., Tignor M., Miller H.L. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 18 p.
- ITDG (2000) Biogas and Biofuels: Technical Brief, http://www. itdg.org/html/technical\_enquiries/docs/biogas\_liquid\_fuels. pdf.
- Jimenez-Gonzalez C., Overcash M. (2000) Life cycle assessment of refineryproducts: review and comparison of commercially available databases, Environ. Sci. Technol. 34, 4789–4796.
- Jolliet O., Müller-Wenk R., Bare J., Brent A., Goedkoop M., Heijungs R., Itsubo N., Peña C., Pennington D., Potting J., Rebitzer G., Stewart M., Udo De Haes H., Weidema B. (2004) The LCIA Midpoint-damage Framework of the UNEP/SETAC Life Cycle Initiative, Int. J. LCA 9, 394–404.
- Jørgensen R., Jørgensen B.J., Nielsen N.E., Maag M., Lind A.M. (1997) N<sub>2</sub>O emission from energy crop fields of *Miscanthus giganteus* and winter rye, Atmos. Environ. 31–18, 2899–2904.
- JRC/EUCAR/CONCAWE (2006) JEC Well-to-Wheels study Version 2b, May 2006, http://ies.jrc.ec.europa. eu/our-activities/support-to-eu-policies/ well-to-wheels-analysis/WTW.html.
- JRC/EUCAR/CONCAWE (2008) JEC Well-to-Wheels study Version 3, year 2008, http://ies.jrc.ec.europa. eu/our-activities/support-to-eu-policies/ well-to-wheels-analysis/WTW.html.
- Jug A., Makeschin F., Rehfuess K.E., Hofmann-Schielle C. (1999) Short-rotation plantations of balsam poplars, aspen and willows on former arable land in the Federal Republic of Germany. III. Soil ecological effects, Forest Ecol. Manag. 121, 85–99.
- Kadam K.L. (2000) Environmental life cycle implications of using bagasse-derived ethanol as a gasoline oxygenate in Mumbai (Bombay), NREL/TP-580-28705, 89 p.
- Kahle P., Beuch S., Boelcke B., Leinweber P., Schulten H.R. (2001) Cropping of *Miscanthus* in Central Europe: biomass

production and influence on nutrients and soil organic matter, Eur. J. Agron. 15, 171–184.

- Kaiser E.A., Kohrs K., Kucke M., Schnug E., Heinemeyer O., Munch C. (1998) Nitrous oxide release from arable soil: importance of N fertilization, crops and temporal variation, Soil Biol. Biochem. 30, 1553–1563.
- Kaylen M., Van Dyne D.L., Choi Y.-S., Blase M. (2000) Economis feasibility of producing ethanol from lignocellulosic feedstock, Bioresource Technol. 72, 19–32.
- Khalifa K. (1999) Analyse du Cycle de Vie. Problématique de l'évaluation des impacts, in: Techniques de l'ingénieur, G 5610, 1999.
- Kim S., Dale B.E. (2002) Allocation procedure in ethanol production system from corn grain. I. System expansion, Int. J. LCA 7, 237–243.
- Kodera K. (2007) Analysis of allocation methods of bioethanol LCA, Internship report, Faculty of Earth and Life Science, MSc in Environmental Resource and Management, Amsterdam, 54 p.
- Kojima M., Johnson T. (2005) Potential for biofuels for transport in developing countries, Energy Sector Management Assistance Programme (ESMAP), The World Bank Group, USA, 182 p.
- Köllner T. (2001) Land use in product life cycles and its consequences for ecosystem quality, PhD thesis.
- Kram J.W. (2007) Waste management commits to landfill gas expansion, in: Biomass Magazine 9/2007, Volume I issue 4: Back to basics, BBI International, 54 p.
- Lal R. (2004a) Soil carbon sequestration to mitigate climate change, Geoderma 123, 1–22.
- Lal R. (2004b) Soil carbon sequestration impacts on global climate change and food security, Science 304, 1623–1627.
- Lal R. (2004c) Agricultural activities and the global carbon cycle, Nutr. Cycl. Agroecosys. 70, 103–116.
- Lang X., Dalai A.K., Bakhshi N.N., Reaney M.J., Hertz P.B. (2001) Preparation and characterization of bio-diesels from various bio-oils, Bioresource Technol. 80, 53–62.
- Larson E.D. (2006) A review of life-cycle analysis studies on lisuid biofuel systems for the transport sector, Energy for Sustainable Development X, 109–126.
- Larsson L., Ferm M., Kasimir-Klemedtsson A., Klemedtsson L. (1998) Ammonia and nitrous oxide emissions from grass and alfalfa mulches, Nutr. Cycl. Agroecosys. 51, 41–46.
- Latner K., O'Kray C., Jiang J. (2006) China, Peoples Republic of Bio-Fuels An Alternative Future for Agriculture 2006, USDA Foreign Agriculture Service, Global Agriculture Information Network (GAIN), Report number: CH6049, 29 p.
- Latner K., Wagner O., Jiang J. (2007) China, Peoples Republic of Bio-Fuels An Alternative Future for Agriculture 2006, USDA Foreign Agriculture Service, Global Agriculture Information Network (GAIN), Report number: CH7039, 14 p.
- Laville P., Hénault C., Renault P., Cellier P., Oriol A., Devis X., Flura D., Germon J.C. (1997) Field comparison of N<sub>2</sub>O measurements using micrometeorological and chamber methods, Agronomie 17, 375–388.
- Lemus R., Lal R. (2005) Bioenergy crops and carbon sequestration, Crit. Rev. Plant Sci. 24, 1–21.
- Levelton (1999) Alternative and Future Fuels and Energy Sources for Road Vehicles, Levelton Engineering Ltd, pre-

pared for Canadian Transportation Issue Table, National Climate Change Process.

- Lewandowski I. Scurlock J.M.O. Lindvall E., Christou M. (2003) The development and current status of perennial rhizomatous grasses as energy crops in the US and Europe, Biomass Bioenerg. 25, 335–361.
- Li C.S. (2000) Modelling trace gas emissions from agricultural ecosystems, Nutr. Cycl. Agroecosys. 58, 259–276.
- Li C.S. (2007) Quantifying greenhouse gas emissions from soils: scientific basis and modeling approach, Soil Sci. Plant Nutr. 53, 344–352.
- Li C.S, Frolking S., Frolking T.A. (1992) A model of nitrous oxide evolution from soil driven by rainfall events: 1. Model structure and sensitivity, J. Geophys. Res. 97, 9759–9776.
- Lo S.C., Ma H.W., Lo S.L. (2005) Quantifying and reducing uncertainty in life cycle assessment using the Bayesian Monte Carlo method, Sci. Total Environ. 340, 1–3; 23–33.
- Luneau G., Fayet S. (2007) Les biocarburants, présentation des principales filières et impacts environnementaux-qualité de l'air, Synthèse bibliographique pour AtmoPACA, Qualité de l'Air, France, mars 2007, 23 p.
- Ma F., Hanna M.A. (1999) Biodiesel production: a review, Bioresource Technol. 70, 1–15.
- Ma Z., Wood C.W., Bransby D.I. (2000a) Soil management impacts on soil carbon sequestration by switchgrass, Biomass Bioenerg. 18, 469–477.
- Ma Z., Wood C.W., Bransby D.I. (2000b) Carbon dynamics subsequent to establishment of switchgrass, Biomass Bioenerg. 18, 93–104.
- MacDonald T., Perez P., Mizutani C., Therkelsen R. (2004) Ethanol Fuel Incentives Applied in the US. Reviewed from California's Perspective, California Energy Commission, Staff report, USDOE, P600-04-001, 36 p.
- Malça J., Freire F. (2006) Renewability and life-cycle energy efficiency of bioethanol and bio-ethyl ertiary butyl ether (bioETBE): Assessing the implications of allocation, Energy 31, 3362–3380.
- Mattsson B., Cederberg C., Blix L. (2000) Agricultural land use in LCA: case studies of three vegetable oil crops, J. Cleaner Prod. 8, 283–292.
- Milà I Canals L., Müller-Wenk R., Bauer C., Depestele J., Dubreuil A., Freiermuth K.R., Gaillard G., Michelsen O. Rydgren B. (2007) Key Elements in a Framework for Land Use Impact Assessment Within LCA, Int. J. LCA 12–1, 5–15.
- Möller R., Pauly M., Hake S. (2006) Cell wall saccharification, Outputs from the EPOBIO project, CPLpress Science Publishers © CNAP, University of York, ISBN 10: 1-872691-04-8, ISBN 13: 978-1-872691-04-6, 67 p.
- Möller R., Pauly M., Hake S., Bowles D. (2007) Crop platforms for cell wall saccharification: lignocellulose feedstock, Outputs from the EPOBIO project, CPLpress Science Publishers © CNAP, University of York, ISBN 10: 1-872691-13-7, ISBN 13: 978-1-872691-13-8, 176 p.
- Moras S. (2007) Analyse comparée du cycle de vie des filières de production d'énergie renouvelable issue de la biomasse, Thèse de doctorat, Gemblou, Faculté universitaire des Sciences agronomiques, 344 p., 80 tabl., 100 fig.
- Moreira J.S. (2000) Sugarcane for energy recent results and progress in Brazil, Energy for Sustainable Development 6, 43–54.

- Mosier A.R., Duxbury J.M., Freney J.R., Heinemeyer O., Minami K. (1996) Nitrous oxide emissions from agricultural fields: Assessment, measurement and mitigation, Plant Soil 181, 95–108.
- Mosier A.R., Kroeze C., Nevison C., Oenema O., Seitzinger S., van Cleemput O. (1998) Closing the global N<sub>2</sub>O budget: nitrous oxide emissions through the Agricultural nitrogen cycle, Nutr. Cycl. Agroecosys. 52, 225–248.
- Murugesan A., Umarani C., Subramanian R., Nedunchezhian N. (2009) Bio-diesel as an alternative fuel for diesel engines - A review, Renew. Sust. Energ. Rev. 13, 653–662.
- Naylor R., Liska A.J., Burke M.B., Falcon W.P., Gaskell J.C., Rozelle S.D., Cassman K.G. (2007) The ripple effect: biofuels, food security, and the environment, Environment 49, 31–43.
- NDRC (2007) China's National Climate Change Programme, prepared under the Auspices of National Development and Reform Commission (NDRC), People's Republic of China, 63 p.
- Nguyen T.L.T., Gheewala S.H. (2008) Fuel ethanol from cane molasses in Thailand: Environmental and cost performance, Energy Policy 36, 1589–1599.
- Nguyen T.L.T., Gheewala S.H., Garivait S. (2007) Energy balance and GHG-abatement cost of cassava utilization for fuel ethanol in Thailand, Energy Policy 35, 4585–4596.
- OECD (2005) Agricultural Policies in OECD Countries: Monitoring and Evaluation 2005, Catalogue n° 512005021P, Paris, DEFRA, UK, 314 p.
- OECD (2006) Agricultural market impacts of future growth in the production of biofuels, Working party on Agricultural Policies and Markets, OECD Directorate for Food, Agriculture and Fisheries Committee for Agriculture, AGR/CA/APM(2005)24/final, 55 p.
- OECD (2008) Environmental performances of agriculture in OECD countries since 1990, ISBN 978-92-64-04092-2, 62 p.
- OECD/FAO (2007) Agricultural Outlook 2007–2016, © OECD/FAO, 88 p.
- Oenema O., Wrage N., Velthof G.L., Van Groenigen J.W., Dolfing J., Kuikman P.J. (2005) Trends in global nitrous oxide emissions from animal production systems, Nutr. Cycl. Agroecosys. 72, 51–65.
- Öhgren K., Bura R., Lesnicki G., Saddler J., Zacchi G. (2007) A comparison between simultaneous saccharification and fermentation and separate hydrolysis and fermentation using steam-pretreated corn stover, Process Biochem. 42, 834–839.
- Öko-Institut (2004) Stoffstromanalyse zur nachhaltigen energetischen Nutzung von Biomasse, Uwe R. Fritsche u.a., Öko-Institut (Projektleitung) in Kooperation mit FhI-UMSICHT, IE, IFEU, IZES, TU Braunschweig, TU München; Endbericht zum Verbundforschungsvorhaben, gefördert vom BMU, Darmstadt usw. (Bezug als pdf unter http://www.oeko. de/service/bio.)
- Oorts K., Merckx R., Gréhan E., Germon J.C., Nicolardot N. (2007) Determinants of annual fluxes of CO<sub>2</sub> and N<sub>2</sub>O in long-term no-tillage systems in northern France, Soil Till. Res. 95, 133–148.
- Parikka M. (2004) Global biomass fuel resource, Biomass Bioenerg. 27, 613–620.
- Parkin T.B. (1987) Soil microsites as a source of denitrification variability, Soil Sci. Soc. Am. J. 51, 1194–1199.

- Parton W.J., Holland E.A., Del Grosso S.J., Hartman M.D., Martin R.E., Mosier A.R., Ojima D.S., Schimel D.S. (2001) Generalized model for NOx and N<sub>2</sub>O emissions from soils, J. Geophys. Res.-Atmos. 106, 17403–17419.
- Parton W.J., Mosier A.R., Ojima D.S., Valentine D.W., Schimel D.S., Weier K., Kulmala A.E. (1996) Generalized model for N<sub>2</sub> and N<sub>2</sub>O production from nitrification and denitrification, Global Biogeochem. Cycles 10, 401–412.
- Pattey E., Edwards G.C., Desjardins R.L., Pennock D.J., Smith W., Grant B., MacPherson J.I. (2007) Tools for quantifying N<sub>2</sub>O emissions from agroecosystems, Agr. Forest Meteorol. 142, 103–119.
- Pelkmans L. Portouli E. Papageorgiou A., Georgopoulos P. (2006) Impact assessment of measures towards the introduction of biofuels in the European Union, PREMIA WP4 report, 133 p.
- Pennington D.W., Potting J., Finnveden G., Lindeijer E., Jolliet O., Rydberg T., Rebitzer G. (2004) Life cycle assessment Part 2: Current impact assessment practice, Environ. Int. 30, 721–739.
- Perlack R.D., Wright L.L., Turhollow A.F., Graham R.L., Stokes B.J., Erbach D.C. (2005) Biomass as Feedstock for a Bioenergy and Bioproducts Inductry: The Technical Feasability of a Bilion-Ton Annual Supply, prepared by Oak Ridge National Laboratory, Tennessee, US-DOE/USDA, 78 p.
- Plassat G. (2005), TOME 1 Les technologies des moteurs de véhicules lourds et leurs carburants, Collection : Données et références, ©ADEME Editions, Paris, 216 p.
- Poiret M. (1996) Evolution des cultures et impacts sur l'environnement, Eurostat. http://ec.europa.eu/agriculture/ envir/report/fr/evo\_cu\_fr/report\_fr.htm. 01/01/2009.
- Pons E. (2008) Sustainable biofuels certification: compatibility of the European Directive on biofuels with WTO trade rules, Ministère de l'Écologie, de l'Énergie, du Développement Durable et de l'Aménagement du Territoire, MPRA, Munich Personal RePEc Archive, Paper No. 16015 posted 01. July 2009, online: http://mpra.ub.uni-muenchen.de/16015.
- Potter C.S., Riley R.H., Klooster S.A. (1997) Simulation modelling of nitrogen trace gas emissions along an age gradient of tropical forest soils, Ecol. Model. 97, 179–196.
- Potting J., Hauschild M. (2005) Background for spatial differentiation in LCA impact assessment – The EDIP2003 methodology, Danish Ministry of the Environment, Environmental Project No. 996 2005, 293 p.
- Potting J., Schöpp W., Blok K., Hauschild M. (1998) Sitedependent life-cycle impact assessment of acidification, J. Ind. Ecol. 2, 63–87.
- Powlson D.S., Riche A.B., Shield I. (2005) Biofuels and other approaches for decreasing fossil fuel emissions from agriculture, Ann. Appl. Biol. 146, 193–201.
- Prasertsri P. (2006) Thailand sugar annual 2006, USDA Foreign Agricultural Service, Gain Report, Global Agriculture Information Network, http://www.fas.usda.gov/gainfiles/200604/ 146187503.docS.
- Prieur A., Bouvart F. (2006) IFP, Rapport de projet ANABIO, PNRB, Tâche 2 : définition du cadre d'étude, Working paper not published, Paris, 29 p.
- Quirin M., Gärtner S.O., Pehnt M., Reinhardt G. (2004) CO<sub>2</sub> Mitigation through Biofuels in the Transport Sector, Status

and Perspectives, Main report, IFEU, Heidelberg, Germany, 66 p.

- Rabl A., Benoist A., Dron D., Peuportier B., Spadaro J.V., Zoughaib A. (2007) How to Account for CO<sub>2</sub> Emissions from Biomass in an LCA, Int. J. LCA 12, 281.
- Radich A. (2004) Biodiesel Performance, Costs, and Use, Analysis report, Energy Information Administration, USDOE.
- Ragauskas A.J., Williams C.K., Davison B.H., Britovsek G., Cairney J., Eckert C.A., Frederick W.J. Jr., Hallett J.P., Leak D.J., Liotta C.L., Mielenz J.R., Murphy R., Templer R., Tschaplinski T. (2006) The Path Forward for Biofuels and Biomaterial, Science 311, 484–489.
- Rajagopal D., Sexton S.E., Roland-Host D., Zilberman D. (2007) Challenge of biofuel: filling the tank without emptying the stomach? Environ. Res. Lett. 2, 1–9.
- Reap J., Roman F., Duncan S., Bras B. (2008) A survey of unresolved problems in life cycle assessment. Part 2: impact assessment and interpretation, Int. J. LCA 13, 374–388.
- Rebitzer G., Ekvall T., Frischknecht R., Hunkeler D., Norris G., Rydberg T., Schmidt W.-P., Suh S., Weidema B.P., Pennington D.W. (2004) Life cycle assessment Part 1: Framework, goal and scope definition, inventory analysis, and applications, Environ. Int. 30, 701–720.
- Recous S. (2001) Dynamics of soil and fertilizer nitrogen in arable systems, paper presented at the International Fertilizer Society at a Symposium in Lisbon, 4 March 2001, Proceedings - International Fertilizer Society, Issue: 463, pp. 1–16.
- Recous S., Machet J.M., Mary B. (1988) The fate of labelled <sup>15</sup>N urea and ammonium nitrate applied to a winter wheat crop. II Plant uptale and N efficiency, Plant Soil 112, 215–224.
- REFUEL (2008) Eyes on the track, mind on the horizon. From inconvenient rapeseed to clean wood: a European roadmap for biofuels, Intelligent Energy Europe, 52 p.
- Reinhardt G., Gärtner S., Retenmaier N., Münch J., Falkenstein E. v. (2007) Screening Life Cycle Assessment of *Jatropha* Biodiesel, IFEU Commissioned by Daimler AG Stuttgart, 62 p.
- REN21 (2008) Renewables 2007 Global Status Report (Paris: REN21 Secretariat and Washington, DC:Worldwatch Institute), Copyright © 2008 Deutsche Gesellschaft für Technische Zusammenarbeit (GTZ) GmbH.
- Ribeiro K.S., Kobayashi S., Beuthe M., Gasca J., Greene D., Lee D.S., Muromachi Y., Newton P.J., Plotkin S., Sperling D., Wit R., Zhou P.J. (2007) Transport and its infrastructure, Climate Change 2007: Mitigation, Contribution of Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Metz B., Davidson O.R., Bosch P.R., Dave R., Meyer L.A. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Richard G., Cellier P. (1998) Effect of tillage on bare soil energy balances and thermal regime: an experimental study, Agronomie 18, 163–180.
- Robertson G.P., Paul E.A., Harwood R.R. (2000) Greenhouse Gases in Intensive Agriculture: Contributions of Individual Gases to the Radiative Forcing of the Atmosphere, Sciences 289, 1922–1925.
- Roby F. (2006) Vers la voiture sans pétrole ? Collection Bulles de Sciences, EDP Sciences, ISBN: 2-86883-874-X.
- Rosegrant M.W. (2008) Biofuels and Grain Prices: Impacts and Policy Responses, Testimony for the U.S. Senate Committee

on Homeland Security and Governmental Affairs, 4 p., http:// www.ifpri.org.

- Roy R.N., Misra R.V., Lesschen J.P., Smaling E.M. (2003) Assessment of soil nutrient balance, Approaches and Methodologies, FAO Fertilizer and plant nutrition bulletin 14, ISBN 92-5-105038-4, Rome, 101 p.
- Russi D. (2008) An integrated assessment of a large-scale biodiesel production in Italy: Killing several birds with one stone? Energy Policy 36, 1169–1180.

Saffih-Hdadi K., Mary B. (2008). Modeling consequences of straw residues export on soil organic carbon, Soil Biol. Biochem. 40, 594–607.

- Schenck R.C., Vickerman S. (2001) Developing a Land Use/Biodiversity Indicator for Agricultural Products LCA's, oral presentation at the International Conference on LCA in Foods, Gothenburg, Sweden, 26–27 April 2001, 6 p.
- SCIO (2007) China's Energy Conditions and Policies; White Paper. Information Office of the State Council of the People's Republic of China, 46 p.
- Seguin B., Arrouays D., Balesdent J., Soussana J.F., Bondeau A., Smith P., Zaehle S., de Noblet N., Viovy N. (2007) Moderating the impact of agriculture on climate, Agr. Forest Meteorol. 142, 278–287.
- Shahid E.M., Jamal Y. (2008) A review of biodiesel as vehicular fuel, Renew. Sust. Energ. Rev. 12, 2484–2494.
- Six J., Ogle S.M., Breidt F.J., Conant R.T., Mosier A.R., Paustian K. (2004) The potential to mitigate global warming with notillage management is only realized when practised in the long term, Global Change Biol. 10, 155–160.
- Skiba U.M., Mctaggart I.P., Smith K.A., Hargreaves K.J., Fowler D. (1996) Estimates of nitrous oxide emissions from soil in the UK, Energ. Convers. Manage. 37, 1303–1308.
- Smeets E.M.W., Faaij A.P.C., Lewandowski I., Turkenburg W.C. (2007) A bottom-up assessment and review of global bioenergy potentials to 2050, Prog. Energ. Combust. 33-1, 56–106.
- Smil V. (1999) Long-range perspectives on inorganic fertilizers in global agriculture, Travis P. Hignett Lecture, International Fertilizer Development Center, Lecture Series LS-2, 3C, 40 p.
- Smith P. (2004) Carbon sequestration in croplands: the potential in Europe and the global context, Eur. J. Agron. 20, 229–236.
- Smith P., Martino D., Cai Z., Gwary D., Janzen H., Kumar P., McCarl B., Ogle S., O'Mara F., Rice C., Scholes B., Sirotenko O. (2007) Agriculture, Climate Change 2007: Mitigation. Contribution of Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Metz B., Davidson O.R., Bosch P.R., Dave R., Meyer L.A. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Smith P., Martino D., Zucong C., Gwary D., Janzen H., Kumar P., McCarl B., Ogle S., O'Mara F., Rice C., Scholes B., Sirotenko O., Howden M., McAllister T., Pan G., Romanenkov V., Schneider U., Towprayoon S., Wattenbach M., Smith J. (2008) Greenhouse gas mitigation in agriculture, Philos. T. Roy. Soc. B 363, 789–813.
- Solomon B.D., Barnes J.R., Halvorsen K.E. (2007) Grain and cellulosic ethanol: History, economics, and energy policy, Biomass Bioerneg. 31, 416–425.

- Sourie J.-C., Tréguer D., Rozakis S. (2005), Ambivalence of biofuel chains in France, in: INRA Sciences Sociales: Research in economics and rural sociology, INRA Social Sciences, Agriculture and Food, Space and Environment, ISSN 1778–4379, Grignon, France, 5 p.
- Soussana J.F., Loiseau P., Vuichard N., Ceschia E., Balesdent J., Chevallier T., Arrouays A. (2004) Carbon cycling and sequestration opportunities in temperate grasslands, Soil Use Manage. 20, 219–230.
- Stanway D. (2008) China needs new policies to kickstart biofuel sector – expert, in: Thomson Financial News, online http://www.forbes.com/afxnewslimited/feeds/afx/2008/ 06/03/afx5073275.html (06.03.08).
- Stead D. (2007) Transport energy efficiency in Europe: Temporal and geographical trends and prospects, J. Transport Geography 15, 343–353.
- Steen B. (1999) A Systematic Approach to Environmental Priority Strategies in Product Development (EPS). Version 2000 a) General System Characteristics. b) Models and Data, Chalmers University of Technology, Centre for Environmental Assessment of Products and material Systems (CPM), Report 1999:4 and 5, Gothenburg, online: http://eps.esa.chalmers.se/download.htm.
- Stehfest E., Bouwman L. (2006) N<sub>2</sub>O and NO emissions from agricultural fields and soils under natural vegetation: summarizing available measurement data and modeling of global annual emissions, Nutr. Cycl. Agroecosys. 74, 207–228.
- Stern N. (2006) Stern Review on the Economics of Climate Change, online: http://www.hm-treasury.gov.uk/stern\_ review\_report.htm.
- Sticklen M. (2007) Spartans tailor maize for cellulosic ethanol, in: Biomass magazine 7/2007: Volume I issue 2: Powering up with poultry litter, BBI International, 50 p.
- Strege J.R. EERC (2007) A road Map for Biofuels Researchpart II, in Biomass Magazine 9/2007, Volume I issue 4: Back to basics, BBI International, 54 p.
- Sutton M.A., Nemitz E., Erisman J.W., Beier C., Butterbach Bahl K., Cellier P., de Vries W., Cotrufo F., Skiba U., Di Marco C., Jones S., Laville P., Soussana J.F., Loubet B., Twigg M., Famulari D., Whitehead J., Gallagher M.W., Neftel A., Flechard C.R., Herrmann B., Calanca P.L., Schjoerring J.K., Daemmgen U., Horvath L., Tang Y.S., Emmett B.A., Tietema A., Peñuelas J., Kesik M., Brueggemann N., Pilegaard K., Vesala T., Campbell C.L., Olesen J.E., Dragosits U., Theobald M.R., Levy P., Mobbs D.C., Milne R., Viovy N., Vuichard N., Smith J.U., Smith P., Bergamaschi P., Fowler D., Reis S. (2007) Challenges in quantifying biosphere-atmosphere exchange of nitrogen species, Environ. Pollut. 150, 125–139.
- Thomsen I.K., Christensen B.T. (2004) Yields of wheat and soil carbon and nitrogen contents following long-term incorporation of barley straw and ryegrass catch crops, Soil Use Manage. 20, 432–438.
- Tolan J.S. (2006) Iogen's demonstration process for producing ethanol from cellulosic biomass, in: Biorefineries-Industrial Processes and Products, Vol. 1, in: Kamm B., Gruber P.R., Kamm M. (Eds.), Wiley-VCH Weinheim, Germany, pp. 193–208.
- Tripartite Task Force (2007) White paper on internationally compatible biofuel standards, Tripartite Task Force Brazil, European Union & United State of America, 94 p.

- Trossero M.A. (2002) Wood energy: the way ahead, Unasylva 211–53, 10 p.
- Tubiello F., Fischer G. (2007) Reducing climate change impacts on agriculture: Global and regional effects of mitigation, 2000–2080, Technol. Forecasting Soc. Change 74, 1030–1056.
- Udo de Haes H.A., Lindeijer E. (2001) The conceptual structure of Life Cycle Impact Assessment. Final draft for the Second Working Group on Impact Assessment of SETAC-Europe (WIA-2), Brussels.
- Udo de Haes H.A, Jolliet O., Finnveden G., Hauschild M., Krewitt W., Müller-Wenk (1999) Best Available Practice Regarding Impact Categories and Category Indicators in Life Cycle Assessment, SETAC-Europe: Second Working Group on LCIA (WIA-2), Int. J. LCA 4, 66–74.
- UN (2006) World population prospects: The 2006 revision, Population database, online: http://esa.un.org/unpp/p2k0data.asp (26.02.2008).
- UNDP (2000) World Energy Assessment: Energy and the challenge of sustainability, World Energy Assessment of the United Nations, UNDP, UNDESA/WEC, New York, 506 p.
- UNEP (2003) Evaluation of Environment Impacts in Life Cycle Assessment, Meeting report, Brussels 29–30 November 1998 and Brighton 25–26 May 2000, UN Publication, ISBN: 92-807-2144-5, 108 p.
- UNFCCC (2006) CDM Executive Board, EB 23 Report, Annex 18, p. 1.
- Ugarte D., Walsh M. (2002) Synergism between Agricultural and Energy Policy: The Case of Dedicated Bioenergy Crops, Agricultural Policy Analysis Center, University of Tennessee, 18 p. online: http://agpolicy.org/ppap/pdf/02/biocrop. pdf.
- US DOE (2006a) Breaking the biological barriers to cellulosic ethanol: a joint research agenda, DOE/SC-0095, US Department of Energy Office of Science and Offie of Energy Efficiency and Renewable Energy, 216 p.
- US DOE (2006b) On the Road to Energy Security, Implementing a Comprehensive Energy Strategy: A Status Report, A message from the Secretary of Energy, USA, 12 p.
- US GAO (2000) Tax Incentives for Petroleum and Ethanol Fuels, GAO/RCED-00-301R. B-286311, United States General Accounting Office/Resources, Community, and Economic Development Division, Washington, DC 20548, September 25, 25 p.
- US Government (2008) figures retrieved from the webpage http://www.whitehouse.gov on the 25/04/2008.
- Van Camp L., Bujarrabal B., Gentile A.-R., Jones R.J.A., Montanarella L., Olazabal C., Selvaradjou S.-K. (2004) Reports of the Technical Working Groups Established under the Thematic Strategy for Soil Protection, EUR 21319 EN/3, 872 p., Office for Official Publications of the European Communities, Luxembourg.
- Van der Drift A., Boerrigter H. (2006) Synthesis gas from biomass for fuels and chemicals, ECN-C—06-001, Report following the workshop "Hydrogen and synthesis gas for fuels and chemicals" organized by IEA bioenergy Task 33 in conjunction with the SYNBIOS conference held in May 2005 in Stockholm, 30 p.
- Velthof G.L., Kuikman P.J., Oenema O. (2002) Nitrous oxide emission from soils amended with crop residues, Nutr. Cycl. Agroecosys. 62, 249–261.

- Von Braun J. (2007) The World Food Situation: New Driving Fores and Required Actions, IFPRI's Bi-Annual Overview of the World Food Situation presented to the CGIAR Annual General Meeting, Beijing, December 3, 2007, 25 p., online: http://www.ifpri.org.
- Wagner-Riddle C., Furon A., Mclaughlin N.L., Lee I., Barbeau J., Jayasundara S., Parkin G., Bertold P. von, Warland J. (2007) Intensive measurement of nitrous oxide emissions from a corn-soybean-wheat rotation under two contrasting management systems over 5 years, Global Change Biol. 13, 1722–1736.
- Weidema B.P., Lindeijer E. (2001) Physical impacts of land use in product life cycle assessment, Technical University of Denmark, 2001.
- Weidema B.P., Wesnaes M.S. (1996) Data quality management for life cycle inventories-an example of using data quality indicators, J. Cleaner Prod. 4, 167–174.
- Wenzel H., Hauschild M., Alting L. (1997): Environmental assessment of products, Volume 1, Methodology, tools and case studies in product development, Chapman and Hall.
- Wiesenthal T., Schade B., Chritidis P., Leduc G., Pelkmans L. (2007) Analysis of biofuel support policies, Proceedings of the 15th Biomass Conference and Exhibition, 7–11 May 2007, Berlin, Germany.
- Worldwatch Institute (2007) Biofuels for Transport: Global Potential and Implications for Energy and Agriculture, prepared by Worldwatch Institute for the German Ministry of Food, Agriculture and Consumer Protection (BMELV) in coordination with the German Agency for Technical

Cooperation (GTZ) and the German Agency of Renewable Resources (FNR), ISBN: 1844074226, Earthscan, London, 452 p.

- Wrage N., Velthof G.L., van Beusichem M.L., Oenema O. (2001) Role of nitrifier denitrification in the production of nitrous oxide, Soil Biol. Biochem. 33, 1723–1732.
- WRI (2007) http://earthtrends.wri.org/maps\_spatial/watersheds/ index.php.
- Wu M., Wang M., Huo H. (2006) Fuel-Cycle Assessment of Selected Bioethanol Production Pathways in the United States, Energy Systems division, Argonne National Laboratory, 65 p.
- Xavier M.R. (2007) The Brazilian sugarcane ethanol experience, Competitive Enterprise Institute CEI Issue Analysis, Advancing Liberty from the Economy to Ecology 3, 13 p.
- Yi I., Itsubo N., Inaba A., Matsumoto K. (2007) Development of the interregional I/O based LCA method considering Region-specifics of indirect effects in regional evaluation, Input-output and hybrid LCA, Int. J. LCA 12, 353–364.
- Zan C.S., Fyles J.W., Girouard P., Samson R.A. (2001) Carbon sequestration in perennial bioenergy, annual corn and uncultivated systems in southern Quebec, Agr. Ecosys. Environ. 86, 135–144.
- Zeman N. (2007) Shell partner to make hydrogen from glycerine, in: Biomass Magazine 8/2007, Volume I issue 3: Learning from first-generation biomass power producers, BBI International, 50 p.

# Agronomic and Physiological Performances of Different Species of *Miscanthus*, a Major Energy Crop

H.W. Zub and M. Brancourt-Hulmel

### Contents

1	Introd	luction	470
2	Crop	Physiology	471
	2.1	Principal Features of a Miscanthus Crop	471
	2.2	Threshold Temperature for Emergence	
		and Thermal Time for the Growth	
		and Development of a Miscanthus Crop	472
	2.3	Radiation-, Water- and Nitrogen-Use	
		Efficiencies	474
3	Above	e-Ground Biomass Production	475
	3.1	Potential for Above-Ground Biomass	
		Production by Different Species	475
	3.2	Components of Above-Ground Biomass	476
4	Respo	nse of Miscanthus to Stress Conditions	477
	4.1	Effect of Chilling Temperatures	477
	4.2	Effect of Frost	478
	4.3	Effect of Water Availability on Biomass	
		Production and Its Components	479
	4.4	Effect of Nitrogen Availability on Biomass	
		and Its Components	481
5	Concl	usions and Prospects	483
Re	ference	es	483

H.W. Zub (🖂)

INRA, USTL UMR 1281, 80203 Estrees-Mons, Peronne, France e-mail: brancour@mons.inra.fr Abstract The European Union recommends the use of lignocellulosic biomass to produce biofuels in order to reduce greenhouse gas emissions. *Miscanthus*  $\times$ giganteus, a C<sub>4</sub> perennial and rhizomatous plant, has been identified as a good candidate for biomass production because of its high potential yield, of up to 49 t DM.ha<sup>-1</sup> for autumn harvest and 26 t DM.ha<sup>-1</sup> for winter harvest, under low input levels. Here, we review current knowledge on the biomass production in Europe of M.  $\times$  giganteus and its two parental species, M. sinensis and M. sacchariflorus, under different stress conditions. This review identifies two key areas where  $M. \times giganteus$  crops could be improved: (i) tolerance to frost during winter or early spring is essential, mainly in Northern Europe, in order to ensure overwintering and protect young shoots following early emergence. Susceptibility to winter frost at temperatures below -3.5°C for rhizomes and -8°C for young shoots of M.  $\times$  giganteus can lead to significant plant losses and lower yields, and (ii) a good water supply is necessary to ensure good establishment rates and satisfactory biomass production. Reductions of up to 84% in above-ground dry matter production because of a lack of water for the autumn harvest, and up to 26% for the winter harvest have been observed. M. sinensis, which displays greater genetic variability than  $M. \times giganteus$ , will provide the necessary genetic resources for frost and water stress tolerance. It is also necessary to either identify genotypes among M. sinensis species that are able to produce an above-ground biomass yield close to the biomass production of M.  $\times$  giganteus under limited water supplies and/or low temperatures, or to generate new interspecific hybrids of M.  $\times$  giganteus with greater tolerance. Particular attention should be paid

to nitrogen response; although no response to nitrogen supply has been observed in M. × giganteus, M. sinensis produces higher levels of biomass with nitrogen inputs.

**Keywords** *Miscanthus*•Biomass production•Chilling temperature • Frost • Nitrogen supply • Water supply • Improvement

#### Abbreviations

DM	dry matter
WUE	water-use efficiency
NUE	nitrogen-use efficiency
RUE	radiation-use efficiency

### **1** Introduction

Energy security and imperatives related to climate change require the large-scale substitution of petroleum-based fuel. Plant-based biofuels will achieve a reduction in greenhouse gas emissions through the fixation of atmospheric carbon in useful biomass (Ceotto, 2008) and the sequestration of carbon in the soil (Benbi and Brar, 2009). European Directive 2003/30/EC promoted the use of renewable sources by defining targets for biofuels in 2010: 5.75% of the fuel consumed in Europe should be produced from biomass. As ethanol production in France, this corresponds to about 2.5% of the total land planted to cereals to enable its production from conventional crops such as sugarbeet and winter wheat (Gurtler, 2007). To reduce this proportion, new "dedicated" energy crops are of particular interest for the production of biofuel using lignocelluloses, a structural material that accounts for much of the mass of plants. One of the benefits of lignocellulosic biofuel is that it reduces greenhouse gas emissions by 85% compared with reformulated gasoline (Wang et al., 2007). Crutzen et al. (2008) found that ethanol produced from corn and sugarcane had a "net climate warming" effect when compared with oil. Another advantage of lignocellolosic biomass is that it can be produced from abundant and diverse raw materials such as wood, grasses or the non-edible parts of plants.



**Fig. 1** *Miscanthus* × *giganteus* during the third year of growth at Estrees-Mons, France (<sup>©</sup> INRA Lille)

Among these different raw materials, a good candidate is miscanthus, a tall, perennial, rhizomatous C<sub>4</sub> grass of the Poaceae Family (Hastings et al., 2008). It produces a large quantity of biomass under low input levels (Clifton-Brown et al., 2004). Its tropical and subtropical genotypes grow to 3 to 4 m when cultivated in Europe, and more in the warm and wet climates of southeast Asia (Fig. 1). Miscanthus rhizomes, or micro-plants, are planted in spring and the canes produced during the summer are harvested annually during the late autumn or winter, following the second or third growing season of the crop. The lifetime of the crop varies from 20 to 25 years (Lewandowski et al., 2003), and the long-term *Miscanthus* plantations enhance soil carbon sequestration (Hansen et al., 2004). Miscanthus spreads naturally via its underground storage organs or rhizomes, but some species can also be seed-propagated. Because miscanthus is propagated vegetatively, the clone is the variety type.

The cultivation of miscanthus in Europe is mainly based on one species,  $M. \times giganteus$ , which is a

natural triploid hybrid between a diploid *M. sinensis* and a tetraploid *M. sacchariflorus* (Greef and Deuter, 1993; Linde-Laursen, 1993). *M.*  $\times$  giganteus production is characterised by low input levels, so that it is a relatively environmentally-friendly crop (Lewandowski et al., 2000). To date, no susceptibility to diseases or pests has been reported by the authorities in the UK (DEFRA, 2007), the European country where the crop is most widely cultivated.

The principal limitations to miscanthus production from  $M. \times giganteus$  are its high establishment costs, its poor overwintering at some sites and the insufficient water supplies available in Southern regions of Europe.  $M. \times giganteus$  has been found to display very little genetic diversity because of its sterility and vegetative propagation. Most of the clones found in this species were descended directly from the 'Aksel Olsen' clone, as shown by isozyme and DNA studies (Vonwuhlisch et al., 1994; Hodkinson et al., 2002). The only exception may be the 'Hervey' clone that came from Japan to the UK in the 1980s (Greef et al., 1997). The small genetic base of  $M. \times giganteus$  explains why the same clone has nearly always been used during agronomy studies or for cultivation. The sterility of  $M. \times giganteus$  is particularly interesting because it prevents putative invasion by the species; on the other hand, it is an issue in terms of improving biomass production and adapting it to a broad range of climatic conditions. Nevertheless, the Miscanthus genus contains more than 20 species, most of them originating from a broad range of geographic regions in Asia, from the sub-tropics to sub-arctic conditions (Numata, 1974, cited by Clifton-Brown and Lewandowski, 2002). Furthermore, the genetic diversity of the M. sinensis pool is relatively high compared with  $M. \times giganteus$  accessions (Greef et al., 1997). M. sinensis is described as being self-incompatible (Greef and Deuter, 1993), which facilitates crossing and genetic studies; it can therefore be exploited to improve its resistance to abiotic factors or its industrial properties.

In order to highlight the need for research on the adaptation of miscanthus to European conditions, the present paper reviews the agronomic and physiological performance, and response to various stresses, of different *Miscanthus* that could be used to produce biomass in Europe. It focuses on the three species studied in Europe:  $M. \times giganteus$ , M. sinensis and M. sacchariflorus. Stress conditions result from an excess or deficit in a physical or chemical environment.

The environmental conditions that affect plant growth include high or low temperatures, excess rainfall, drought, or inadequate soil mineral nutrients such as nitrogen. Depending on when these environmental factors occur, their intensity and duration can reduce plant vitality and cause damage. The paper is presented in two sections; the first is devoted to a description of the plant, with particular emphasis on its agronomic potential and the variations affecting the production of biomass production by different *Miscanthus* species, while the second concerns the influence of stress conditions on plant traits.

### 2 Crop Physiology

Most agronomic and physiological research has focused on one clone of M. × giganteus. This section describes the life cycle of miscanthus, its requirements for establishment and growth and the efficiencies of the crop in terms of radiation, water and nitrogen (N) use. References to *M. sinensis* or *M. sacchariflorus* are mentioned when available.

# 2.1 Principal Features of a Miscanthus Crop

To establish a crop, the optimum planting date is dependent on the plant material. Whereas rhizomes can be planted from March to May (depending on the climate), micro-plants or plants established in pots should be planted later (late April-May) in order to avoid late frosts and improve establishment rates (Christian and Haase, 2001). In general, the irrigation of newlyplanted rhizomes seems to improve the establishment rate under drier conditions, particularly in Southern Europe. Because of the high production cost of rhizomes or micro-plants, the optimum crop density is very low as the plant is able to produce several stems. The usual density is 2 plants.m<sup>-2</sup>, but this can be increased to 5 plants.m<sup>-2</sup>, the level studied in Europe (Lewandowski et al., 2000). However, Christian et al. (2009) recommended planting at least 14% extra rhizome fragments when using 5-year-old rhizomes in order to attain the required plant density.

After emergence, tillering, corresponding to the number of productive shoots per plant, increases rapidly during May, June and July (e.g. up to 40 stems.plant<sup>-1</sup> for M. × giganteus during a UK experiment, Bullard et al., 1997). Rhizome-propagated plants may have fewer but stronger shoots than micropropagated plants (Lewandowski, 1998). During the growing period, the number of productive shoots falls to 25 stems.plant<sup>-1</sup>. This phenomenon can be compared to tillering regression in wheat, barley or forage grasses (Aspinal, 1961; Thorne, 1962; Gillet, 1980). The youngest tillers decline while the oldest continue to grow in August, September or even October, depending on the climate and time elapsing between emergence and flowering.

During the growing period, leaf area development follows a rapid, one-peak curve. In the second year following establishment, it reaches a maximum leaf area index (LAI) of 6-10, depending on treatments and climate. Maximum LAI values have been observed during the flowering phase, after which the canopy starts to senesce (Cosentino et al., 2007). A mature stand of M.  $\times$  giganteus is able to intercept around 90% of useful photosynthetic radiation when the LAI reaches 3.2 (Clifton-Brown et al., 2000). A similar value was found for *M. sinensis* Goliath (Vargas et al., 2002). The light extinction coefficient (k) through the leaf cover of the crop provides a measurement of the absorption of light by leaves:  $M. \times giganteus$  achieves between 0.56 and 0.68 according to some authors (Cosentino et al., 2007; Clifton-Brown et al., 2000, respectively), whereas M. sinensis Goliath reaches 0.66 (Vargas et al., 2002). Most of these values are close to that of 0.67 reported for maize (Clifton-Brown and Jones, 1997).

The end of the growing period coincides with a drop in temperatures, and full senescence can occur with the first frost, as observed in the autumn in Denmark (Christian and Haase, 2001). However, the oldest leaves towards the base of stems start to senesce earlier. At the end of the growing season, nutrients and photosynthates are remobilised from the stems and leaves to the rhizomes (Christian and Haase, 2001). In *M. sinensis*, this accumulation is synchronised with flowering from July to October, depending on the genotype (Stewart et al., 2009). All stems left standing gradually dry out during the winter, until February/March when the crop is ready for winter harvest.

# 2.2 Threshold Temperature for Emergence and Thermal Time for the Growth and Development of a Miscanthus Crop

The threshold temperature corresponds to the temperature at which the plant can start to grow. Thermal time, which is the cumulated value of degree-days from emergence to flowering time, is often used in crop physiology to estimate or predict the length of different developmental phases under cultivated conditions. The degree-day is a function of baseline temperature. For *Miscanthus*, the threshold and baseline temperatures vary between studies (Clifton-Brown and Jones, 1997; Farrell et al., 2006).

Clifton-Brown and Jones (1997) demonstrated the growth of shoots in a controlled environment at temperatures between 6°C and 20°C, although the extension rate was lower at 6°C than at 15°C or 20°C. Price et al. (2004) concluded that the threshold temperature for growth was 6°C because  $M. \times giganteus$  could grow at this temperature. When comparing different species, Koike et al. (1975) cited by Stewart et al. (2009) observed that a daily mean air temperature of 7°C was necessary to initiate growth of the above-ground organs of M. sinensis. Farrell et al. (2006) studied the effect of temperature on shoot emergence in four miscanthus genotypes: a clone of *M. sacchariflorus*, a clone of *M.*  $\times$  giganteus and two hybrids of *M. sinensis*. Shoot production was markedly influenced by temperature in all genotypes. The rate of emergence increased in line with temperature (Table 1). Genotypes with a lower threshold temperature for emergence were thus able to emerge earlier. Similarly, the thermal responses for emergence in M.  $\times$  giganteus and M. sacchariflorus (slope of regression fits of 0.01 and 0.008, respectively) were lower than those of *M. sinensis* hybrids (between 0.011 and 0.014). Higher emergence rates were associated with both a shorter thermal requirement for emergence and earlier emergence. Farrell et al. concluded as to the existence of genetic variability regarding the threshold temperature required for emergence. Hybrids of M. sinensis had a lower threshold temperature for emergence than M.  $\times$  giganteus and M. sacchariflorus (Table 1).

Both Clifton-Brown et al. (2000 and 2004) and Hastings et al. (2008) preferred to use 10°C as the

	S	hoot emergence ra	te	Leaf extension rate			Plant extension rate				
	Þ.	est-fit regression		best-fit regression			best-fit regression				
	ш	nodel	aı	model	$a_2$	$\mathbf{b}_2$	model	<b>a</b> <sub>3</sub>	$\mathbf{b}_3$	$c_3$	$d_3$
M. sacchariflorus	lii	near	0.008	exponential	0.0007	0.4458					
$M. \times giganteus$	у	$=a_1x+b_1$	0.01	$y = a_2 * e^{b_3}$	0.0032	0.354					
M. sinensis Hybrid 6	(7)	2)	0.011	(2)	0.0039	0.3485					
M. sinensis Hybrid 9			0.014	~	0.0262	0.2434					
M. condensatus $(n = 1)$	mean			$y = a_2 * PER$	0.72			-0.0002	0.0107	-0.0645	0.2070
<i>M.</i> floridulus $(n = 2)$	min			(1)	0.81			-0.0002	0.0095	-0.0574	0.1443
M. saccharifer $(n = 2)$	max				0.88		third-order	-0.0006	0.0242	-0.1800	0.4225
<i>M. sacchariflorus</i> $(n = 3)$					0.83		polynomial	-0.0002	0.0083	-0.0212	-0.0323
$M. \times giganteus (n = 2)$					0.85		$y = a_3 x^3 + b_3 x^2 +$	-0.0002	0.0084	-0.0099	-0.0990
					0.83		$c_3x + d_3(1)$	-0.0001	0.0074	-0.0237	0.0160
M. sinensis $(n = 21)$					0.69			-0.0004	-0.0034	-0.1060	-0.2950
					0.98			0.0002	0.0164	0.0733	0.3050
M. undine $(n = 1)$					0.86			0.0001	-0.0001	0.0534	-0.2130

Table 1 Different relationships between temperature ranges and shoot emergence, and plant and leaf extension rates as a function of different miscanthus species. Data from

baseline temperature in their model to predict potential yield without water stress (Clifton-Brown et al., 2000) or under water stress (Hastings et al., 2008). Indeed, Clifton-Brown et al. (2000) showed a better correlation between cumulative degree-days and leaf area index using a threshold temperature of  $10^{\circ}$ C rather than  $6^{\circ}$ C. It is necessary to carry out similar experiments with *M. sinensis* in order to determine the threshold temperature and thus compare the thermal times of different *Miscanthus* crops.

Applying a threshold temperature of 10°C for three species, the thermal time required to reach peak heading time was estimated at around 900°C for M. sinensis grasses in Japan (Shoji et al., 1990). According to Lewandowski and Clifton-Brown et al. (2000), a thermal time of 1800°C accurately predicted the time for flowering of the only  $M. \times giganteus$  clone studied in Europe. Numata and Mitsudera (1969) cited by Stewart et al. (2009) reported a thermal time of 1500-4000°C for the flowering of M. sinensis grasses in Japan, and this was equivalent to a vegetative growth period of 6-9 months. Although one study with several M. sinensis genotypes was carried out at five sites in Europe (Clifton-Brown et al., 2001a), no data are available on the variations in the thermal time for growth of this species in Europe.

Clifton-Brown et al. (2001a) observed variations in flowering date among genotypes and across countries. The induction of the flowering is dependent on day length and variations are observed among the species. These can be related to the geographical origin of genotypes (Lewandowski et al., 2003). Experiments under artificial light conditions have been carried out to induce flowering. A short-day light period may be necessary to induce the flowering *of M. sacchariflorus* (Deuter, 2000). By contrast, day length may be less important than cumulative degree-days in *M. sinensis*. For instance, a particular genotype of *M. sinensis* flowers earlier in Portugal than in Sweden because it receives its sum of degree-days for flowering more rapidly (Clifton-Brown et al., 2001a).

# 2.3 Radiation-, Water- and Nitrogen-Use Efficiencies

Because of its  $C_4$  photosynthetic pathway and perennial rhizome, miscanthus displays quite a good combination of radiation-, water- and nitrogen-use efficiencies for biomass production (Lewandowski et al., 2000; Heaton et al., 2004; Lewandowski and Schmidt, 2006). Several studies have been carried out to determine these efficiencies in one clone of  $M. \times giganteus$ , and genetic variability has been reported with respect to the radiation- and water-use efficiencies of other *Miscanthus* species.

A linear relationship was found for miscanthus between cumulated above-ground dry biomass (or dry matter, DM) and global intercepted solar radiation (Clifton-Brown et al., 2000). Under water and nitrogen supplies, the radiation-use efficiency (RUE) of  $M. \times giganteus$  reached 4.1 g DM.MJ<sup>-1</sup> in France and the USA (Tayot et al., 1995; Heaton et al., 2008). Lower values of between 1.9 and 2.4 g DM.MJ<sup>-1</sup> were observed, but under conditions of water stress (Bullard et al., 1997; Price et al., 2004; Clifton-Brown et al., 2000) or using a different method to calculate RUE values (Cosentino et al., 2007).

As for water-use efficiency (WUE), studies showed that an adult M. × giganteus stand reached between 9.1 and 9.5 g DM.1<sup>-1</sup> in the UK (Beale et al., 1999) and between 6 and 10 g DM.1<sup>-1</sup> in France (Cadoux et al., 2008). Lower values were found in a Mediterranean environment, with a negative correlation of -0.87 between water availability and WUE (Cosentino et al., 2007). Beale et al. (1999) showed that irrigation during dry periods reduced the WUE of the crop by 15%, with a higher water consumption of 45%.

The average nitrogen-use efficiency (NUE) of above-ground M. × giganteus biomass was calculated at about 277 kg DM.kg<sup>-1</sup> of the N supply (total available nitrogen including soil N and N supply not being evaluated), but considerable variability was observed, ranging from 143 kg DM.kg<sup>-1</sup> in Ireland to 613 kg DM.kg<sup>-1</sup> in Portugal (Christian and Haase, 2001). The NUE calculated from the increase in biomass at different N input levels decreased continuously with an increasing N supply (Lewandowski and Schmidt, 2006; Cosentino et al., 2007). Furthermore, a 9% reduction in NUE was observed with an increase in the water supply (Cosentino et al., 2007).

Genotypic variability has been studied with respect to RUE and WUE, but not NUE. In field conditions under nitrogen supply and without irrigation, Jorgensen et al. (2003a) and Vargas et al. (2002) showed higher RUE values for *M. sinensis* (average: 1.62 g DM.MJ<sup>-1</sup>) than for hybrids between *M. sinensis* (with an intermediate RUE) and *M. sacchariflorus*. Under water stress conditions, RUE values for hybrids between *M. sinensis* and *M. sacchariflorus* were close to those of *M.* × *giganteus*. By contrast, under controlled conditions, Clifton-Brown et al. (2000) demonstrated similar WUE values for young shoots of *M.* × *giganteus* and *M. sinensis* (2 g DM.1<sup>-1</sup>), but the highest value was seen in *M. sacchariflorus* (3.8 g DM.1<sup>-1</sup>).

Several studies comparing the RUE (Beale and Long, 1995; Heaton et al., 2008), NUE (Boehmel et al., 2008; Lewandowski and Schmidt, 2006) and WUE (Beale et al., 1999) of miscanthus with other energy crops concluded as to the superiority of miscanthus efficiencies over other species. Firstly, Beale and Long (1995) and Heaton et al. (2008) found 50% higher RUE values for *Miscanthus*  $\times$  giganteus than for two other rhizomatous perennial crops: Spartina cynosuroides and switchgrass. Secondly, Boehmel et al. (2008) compared the nitrogen-use efficiency of different annual and perennial energy cropping systems and concluded that miscanthus had a higher NUE value of 526 kg DM.kg<sup>-1</sup> when compared with the NUE of energy maize (65 kg DM.kg<sup>-1</sup>). Lewandowski and Schmidt et al. (2006) found the same superiority of miscanthus NUE values over Reed canary grass and triticale, which displayed a stronger decrease in NUE values with nitrogen supply than miscanthus. Finally, Beale et al. (1999) showed than Miscanthus had a WUE value that was 35% to 55% higher than Spartina cynosuroides depending on water supply (rainfed and irrigated, respectively).

### **3 Above-Ground Biomass Production**

Because miscanthus is a perennial grass, the first year concerns establishment of the crop, above-ground biomass production only being possible during subsequent years. The above-ground biomass potential of different species in Europe is discussed first of all, and then the components of this biomass are described.

# 3.1 Potential for Above-Ground Biomass Production by Different Species

The lifetime of the crop is estimated at between 20 and 25 years (Lewandowski et al., 2003), during which miscanthus biomass is produced during two phases: a yield-building phase, which in M. × giganteus lasts

for two to five years, depending on climate and plant densities, and a plateau phase where the yield is maintained (Clifton-Brown et al., 2000, 2001b; Christian et al., 2008). Ceiling yields are attained more rapidly in warmer climates than in cooler climates, especially when crop water supplies are not limiting. Trials in Southern regions have shown that the plants tend to mature more rapidly than at Northern latitudes. The time required to reach the plateau phase is probably species-dependent, but no data are available.

Yield is very low during the first year (less than 10 t.ha<sup>-1</sup> for M. × giganteus) but these figures are usually not known as the grass is not harvested. However, yields in the autumn of the first and second years are a good indicator of those for the third year. Clifton-Brown et al. (2001a) studied the performance of 15 *miscanthus* crops belonging to the M. × giganteus species (4 clones), M. sacchariflorus (1 clone), M. sinensis species (5 clones) and hybrids between M. sacchariflorus and M. sinensis (5 clones), at five sites in Europe during the first three years after establishment and observed a higher and significant correlation of 0.81 between the third-year and second-year yields, this correlation being 0.56 between the third-year and first-year yields.

During subsequent years, peak yields are obtained in the autumn, at the full plant flowering phase (Cosentino et al., 2007) and then decline through the winter due to leaf loss. Harvestable yields in the spring are 27%-50% lower than in the autumn (Clifton-Brown et al., 2001b; Cadoux et al., 2008; Himken et al., 1997; Jorgensen et al., 2003b; Ritchter et al., 2008). Jorgensen et al. (2003b) found similar losses of above-ground dry matter yields that reached 40% and 35% during the second and third years of growth when harvest was delayed from the autumn to the spring in one M. sacchariflorus, five clones of M. sinensis and five hybrids of M. sinensis. Greater above-ground dry matter yield losses were observed for hybrids of M. sinensis than with M. sinensis itself (48% and 35%, respectively, during the second year of growth and 45% and 29% during the third year of growth; Jorgensen et al., 2003b).

A review of autumn yields in Europe showed that the maxima from M. × giganteus were obtained in France in Lusignan (44 °N, 1 °E) and Grignon (48 °N, 2 °E) under irrigated and fertilised conditions: being, respectively, 49 and 42 t DM.ha<sup>-1</sup> (Clifton-Brown et al., 2004). These yields ranged from 15 to 25 t.ha<sup>-1</sup> without irrigation. For winter yields of M. × giganteus, the highest non-irrigated yields were found to be 15-19 t DM.ha<sup>-1</sup> during the trials performed by the Miscanthus Productivity Network that involved 15 European sites. Yields ranged from 7 to 26 t DM.ha<sup>-1</sup> following the third growing season, with some of the trial crops being irrigated (Clifton-Brown et al., 2001b). Outside this network, higher productivity was reported in Central and Southern Europe, but irrigation was always required (Clifton-Brown et al., 2001b; Cadoux et al., 2008).

In order to compare several species, a threeyear study was carried out to evaluate the biomass production potential of four acquisitions of M.  $\times$ giganteus, one clone of M. sacchariflorus, five clones of M. sinensis and five hybrids of M. sinensis (Clifton-Brown et al., 2001a). During the three years of production, the autumn yields demonstrated the higher productivity of M.  $\times$  giganteus when it was able to survive the winter. Apart from  $M. \times giganteus$ , higher yields were reported for *M. sinensis* hybrids, the highest being observed in Germany with mean yields of 11.1 t.ha<sup>-1</sup> for *M. sinensis* and 15.1 t.ha<sup>-1</sup> for hybrids. The mean yields of these hybrids could be between 6% and 90% higher than those of M. sinensis, depending on the country. Jezowski (2008) concluded as to the superiority of a hybrid between M. sinensis  $(2\times)$  and M. sacchariflorus  $(4\times)$  by comparison with a hybrid between *M. sinensis*  $(2\times)$  and *M. sacchariflorus*  $(2\times)$ or between two *M. sinensis*  $(2\times)$ : mean yields reached 20.5, 14.9 and 9.8 t DM.ha<sup>-1</sup>, respectively.

# 3.2 Components of Above-Ground Biomass

These differences among genotypes and regions for above-ground biomass production could be explained by the difference in the duration of the growing period. Tiller densities, plant height and the proportion of leaves and stems were also key elements explaining the variability in biomass production among species. The next section focuses on above-ground biomass production.

#### 3.2.1 Length of Growing Period

Above-ground biomass production is dependent on the duration of the growth period. After the first year, the

start of the growing season is determined by the last spring frost, and the end of the growing season is determined by flowering or the first autumn or winter frost, depending on the harvest date or location.

In European environments, shoot emergence occurs between late March and late April, depending on the climate and the temperature required for the emergence of each genotype (Farrell et al., 2006) (see Sect. 2.2). Because *M. sinensis* needs fewer cumulative degreedays than M. × giganteus and *M. sacchariflorus* prior to emergence (60 vs. 90 and 120 degree-days, respectively), its growing season starts earlier.

The growing season cycle ranges from 6 to 9 months, depending on the genotype. *M. sinensis* genotypes, which flower rapidly, display a shorter growing season than hybrids of *M. sinensis* (which flower later), or *M.* × giganteus, which is generally unable to flower under Northern European conditions (Clifton-Brown et al., 2001a). During the growing season, *M. sinensis* and *M.* × giganteus normally produce new stems throughout the vegetation period (Lewandowski et al., 2003; Stewart et al., 2009) while *M. sacchariflorus* forms about 80% of its stems in the spring.

#### 3.2.2 Plant Morphology

Jezowski (2008) studied two  $M. \times giganteus$ , two hybrids of *M. sinensis*  $(2\times)$  and *M. sacchariflorus*  $(2\times)$ , and two hybrids between two *M. sinensis*. Pooling of their observations revealed a good correlation between yield and shoot numbers and between yield and tuft diameter in 1-year-old plants (0.78 and 0.79, respectively), the mean of productive shoots per plant being 6.6. Stronger correlations were observed during subsequent years (more than 0.90 for 2- and 3-year-old plants; with mean numbers of productive shoots of up to 9.6 and 18.6, respectively), and yield was also correlated with plant height (0.67 in the second year and 0.91 in the third year). Angelini et al. (2009) obtained a correlation of 0.75 between stem numbers and the dry yield of above-ground biomass when they pooled their observations on  $M. \times giganteus$  over a 10-year period.

Stems make a considerable contribution to aboveground biomass, but this is dependent on harvest date. In  $M. \times giganteus$ , stems constitute between 67% and 75% of the standing crop biomass in the autumn, depending on crop age (Tayot et al., 1995; Bullard et al., 1997; Ercoli et al., 1999) and up to 77% or 98% in the winter, depending on location (Tayot et al., 1995; Cosentino et al., 2007; Christian et al., 2008). In *M. sinensis*, similar percentages of stems, between 57% and 78%, have been observed in the autumn (Kalembasa et al., 2005, Jorgensen et al., 2003b), and up to 94% in the winter (Jorgensen et al., 2003b).

## 4 Response of Miscanthus to Stress Conditions

Few studies have been devoted to stress conditions, the important one being the European Miscanthus Improvement Project which was designed to broaden the genetic base, test genotypes and develop breeding methods (Lewandowski and Clifton-Brown, 2000). It contributed to developing new screening techniques to determine genotypic variability for traits such as the response to effects of low temperatures, frost tolerance, mineral content and biomass yield (Lewandowski and Clifton-Brown, 2000). This project identified characteristics related to low-temperature growth and frost tolerance as being crucial to the improvement of miscanthus (Jorgensen and Muhs, 2001). Another project, the Miscanthus Productivity Network, studied the limitations of low temperatures and other abiotic factors on the growth of  $M. \times giganteus$  under European climatic conditions (Walsh, 1998).

Studies on the use of environmental resources have focused on the effect of water availability (Christian and Haase, 2001; Clifton-Brown et al., 2002; Cosentino et al., 2007) and nitrogen availability (Lewandowski et al., 1995; Lewandowski and Kicherer, 1997; Lewandowski and Kauter, 2003; Christian et al., 2006; Lewandowski and Schmidt, 2006; Christian et al., 2008).

The effects of cold temperatures and frost are presented first of all, followed by the effects of water and nitrogen availabilities.

### 4.1 Effect of Chilling Temperatures

Several important air temperature effects on miscanthus growth have been distinguished. Air temperature affects canopy development through organ growth and expansion rates (Clifton-Brown and Jones, 1997; Clifton-Brown et al., 2001b), as well as canopy functioning through photosynthesis (Beale et al., 1996; Weng and Ueng, 1997; Wang et al., 2008a, b).

#### 4.1.1 Effect on Leaf Expansion Rates

Thermal responses affecting the leaf expansion rate have been reported with respect to 32 genotypes grown under controlled conditions (Clifton-Brown and Jones, 1997). After the plants were kept in cabinets at  $20 + / - 3^{\circ}C$  with optimum supplies of water and nutrients, they were then subjected to repeated sequences of stepped temperature treatments during which they were exposed for 18 h to 20°C and for 11.5 h to 15, 10, 7.5 and 5°C, successively. Plant extension rates were measured mechanically by auxonometers attached to the youngest visible expanding leaf on a miscanthus plant. These auxonometers measured a combination of leaf and internodal extension. Leaf extension was estimated using ruler measurements of the leaf attached to the auxonometer. Leaf extension accounted for 83% of plant extension. The relationship between the temperature range and the plant extension rate fitted well with a third-order polynomial (Clifton-Brown and Jones, 1997). Farrell et al. (2006) determined the best fit relationship using an exponential regression model for the leaf extension rate (Table 1).

From the model developed by Clifton-Brown and Jones (1997), the estimated ratio between expansion rates at 10°C and 20°C varied among the genotypes and ranged from 3 to 4.7. All genotypes displayed predicted leaf extension at the lower temperature of 5°C that was always less than 0.1 mm.h<sup>-1</sup>, except for two genotypes with a predicted rate of 0.19 mm.h<sup>-1</sup> and two others which had a leaf extension rate of 0.13 mm.h<sup>-1</sup> (calculated from Clifton-Brown and Jones, 1997). However, these advantages at the lower temperature were lost at the higher temperature.

During the same experiment, simulation with a simple growth model showed that genotypes with lower threshold temperatures for leaf expansion were unlikely to display greater productivity than genotypes with more rapid expansion rates at temperatures higher than 10°C (Clifton-Brown and Jones, 1997; Jorgensen and Muhs, 2001). Some genetic variability existed among genotypes with lower threshold temperatures for leaf expansion and others with a more rapid leaf extension rate when the air temperature was higher than 10°C. Improved biomass yields could thus be expected if these two traits were combined at the optimum level.

#### 4.1.2 Effect on Light Interception and Photosynthesis

Final crop yields are closely linked to canopy photosynthesis during the growing season, which itself is dependent on the amount of radiation intercepted by leaves and the photosynthetic activity of individual leaves. Leaf area development and radiation-use efficiency have already been discussed in Sects. 2.1 and 2.3.

Weng and Ueng (1997) studied the optimum temperature for photosynthesis in *Miscanthus* in Taiwan and explained that it varied in line with habitats: it was higher than 35°C for lowland clones, which is close to the optimum temperature for maize and sugarcane (Yan and Hunt, 1999), but was 28°C for clones at altitudes above 2550 m.

In contrast to related  $C_4$  species such as maize,  $M. \times giganteus$  displays a notable ability to maintain high photosynthetic productivity at low temperatures of 14°C or 10°C (Beale et al., 1996; Naidu et al., 2003; Farage et al., 2006). The molecular mechanism underlying this tolerance remains unclear.  $M. \times giganteus$ maintains a high rate of CO<sub>2</sub> uptake in order to increase its utilisation of absorbed light (Beale et al., 1996), and increases non-photochemical quenching (Farage et al., 2006) to prevent the marked reduction of CO<sub>2</sub> assimilation observed in chilling-intolerant C<sub>4</sub> species.

Nevertheless, the cold tolerance of C<sub>4</sub> photosynthesis in M. × giganteus may be related to the higher levels or activity of two major photosynthesis enzymes: Rubisco and Pyruvate Phosphate DiKinase (PPDK) (Sage and Kubien, 2007). Although no difference in Rubisco content and activity has been observed between M. × giganteus plants grown under cold and warm conditions (Wang et al., 2008a), the rise in PPDK content and activity per unit leaf area in the cold-grown genotype was related to the recovery and maintenance of photosynthetic activity (Wang et al., 2008b). Naidu et al. (2003) found a 28% increase in PPDK levels in cold-grown miscanthus.

#### 4.2 Effect of Frost

Miscanthus may encounter two types of frost temperatures in Europe: negative temperatures during the winter and late frosts during the spring. The former may damage young plantings in Northern Europe and constitute an obstacle to crop establishment; in cooler areas, late frosts can damage newly-expanded leaves during the first year of the crop or just after the regrowth of older plantings.

A capacity for winter survival has been described as the principal obstacle to the establishment of  $M. \times giganteus$  crops (Christian and Haase, 2001), particularly in Northern Europe (Denmark and Sweden), where no plants were able to regrow after the first winter (Lewandowski and Clifton-Brown et al. 2000). Similar mortality rates were observed in Ireland, the Netherlands, Belgium and Germany, as reported by Christian and Haase (2001). During these trials, plants started to grow during the spring after planting, but the first shoots produced were killed by late frosts and the rhizomes were unable to sprout again. The authors suggested that the winter survival of M.  $\times$  giganteus was mainly dependent on the generation of sufficient metabolic reserves in the rhizome system to enable the emergence of sufficient shoots the next spring (Christian and Haase, 2001; Schwarz et al., 1994).

Christian and Haase (2001) suggested that winter survival might also be related to dormancy during frosty periods: frost tolerance may only develop in these organs once dormancy has been induced during the autumn or early winter. Rhizomes and appending buds can be expected to have a higher frost tolerance than overwintering organs. However, a study on the frost tolerance of buds and rhizomes compared with that of leaves showed that this was not the case (Clifton-Brown and Lewandowski, 2000b; Jorgensen and Muhs, 2001; Farrell et al., 2006). This finding may have been due to the protection and isolation offered by the typical micro-structure of plant molecules and the surrounding soil, rather than to the inherent frost tolerance of rhizomes.

Plazek et al. (2009) tested the effects on metabolic activity and rhizome regrowth of exposing leaves to a frost of up to  $-8^{\circ}$ C or  $-15^{\circ}$ C following a 14-day acclimation period at 12°C and a 21-day period at 5°C. Although no frozen plants could produce new shoots, the rhizomes displayed some metabolic vitality

	Leaves (Farrell et al.,	2006)	Rhizomes (Cli Lewandowski	ifton-Brown and et al., 2000b)
	LT <sub>50</sub> (°C)	MC (%)	LT <sub>50</sub> (°C)	MC (%)
M. sacchariflorus	-7.5 ab	87 a	-3.5 b	73 a
$M. \times giganteus$	-8 b	85 a	-3.5 b	76 a
M. sinensis Hybrid 6	6 a	85 a	6 a	68 b
M. sinensis Hybrid 9	-9 c	80 b	-4.5 ab	71 b

**Table 2** Coincidence between the lowest lethal temperature ( $LT_{50}$ ) and lowest moisture content (MC) of leaves and rhizomes in four genotypes of miscanthus but no correlation between  $LT_{50}$  of leaves and  $LT_{50}$  of rhizomes

LT<sub>50</sub>, the lethal temperature at which 50% of shoots (or rhizomes) were considered as killed when the leaves were totally brown. MC, moisture content. Different letters indicate significant differences calculated from the least significant difference (P < 0.05).

after a 5-week regrowth period at 20°C. The authors concluded that frost susceptibility was due to the frost sensitivity of shoot apical meristems.

In terms of species variability, M. sinensis might be expected to be more frost-tolerant than M. sacchariflorus because in Asia, M. sacchariflorus is only found in warm regions, whereas M. sinensis is more ubiquitous (Clifton-Brown and Lewandowski, 2000b). Farrell et al. (2006) compared these species with  $M. \times giganteus$  in terms of cold tolerance at the shoot level (young shoots of 3-4 leaves). Their screening method consisted of growing plants in cabinets at  $20 + -3^{\circ}C$  for 21 days with optimum supplies of water and nutrients. The plants were acclimated at  $12 + 2^{\circ}C$ for 8 days before their leaves were exposed to frost temperatures descending from  $-2^{\circ}C$  to  $-10^{\circ}C$  at a time-step of 2°C every 3 hours. Before this frost treatment, leaves were sampled from each genotype to determine their moisture content. Any frost damage was assessed 7 days later. The lethal temperature at which 50% of the shoots were killed ( $LT_{50}$ ) was estimated for each genotype. The same experiment was carried out on the same genotypes as those studied by Clifton-Brown and Lewandowski (2000b) in order to determine the  $LT_{50}$  of rhizomes.

*M. sinensis* hybrids were found to have significantly lower rhizome moisture contents than *M.* × *giganteus* and *M. sacchariflorus*. These differences coincided with the lower LT<sub>50</sub> of *M. sinensis* (Clifton-Brown and Lewandowski, 2000b; Farrell et al., 2006). Jorgensen et al. (2003b) found a higher rhizome moisture content in *M.* × *giganteus* (none of which survived the first winter) than in *M. sinensis* and most *M. sinensis* hybrids which displayed survival rates of up to 85–99%. However, the degree of resistance to negative temperatures differed in rhizomes and shoots. *M. sinensis* H6 had the lowest  $LT_{50}$  value for rhizomes while *M. sinensis* H9 had the lowest  $LT_{50}$  for shoots (-6°C and -9°C, respectively). Shoot and rhizome frost tolerance were not correlated in these studies (Table 2). Tissue moisture content is considered to be one of the best characteristics for use during plant screening programmes with the accumulation of osmotically active compounds during cold acclimation (Lewitt, 1980). However, no correlations with frost tolerance were found for either leaves or rhizomes (Clifton-Brown and Lewandowski, 2000b; Farrell et al., 2006).

# 4.3 Effect of Water Availability on Biomass Production and Its Components

Increased productivity will result in higher water demand, so that water may become a limiting factor to both crop productivity and the economic viability of the crop. Because of the climatic differences between Northern and Southern Europe, it is therefore necessary to select potential energy crops from species with a high water-use efficiency (Long and Beale, 2001), particularly when adapting the species to waterlimited areas. The identification of drought-tolerant genotypes that can produce more biomass under water stress conditions remains an essential component in the improvement of miscanthus (Clifton-Brown and Lewandowski, 2000a).

The effects of water availability in the field are described with respect to M. × giganteus, particularly



Above-ground biomass yield (t DM.ha<sup>-1</sup>)

Location and age of the crop

**Fig. 2** Increase in dry matter content of above-ground biomass under a water supply in M. × giganteus. Values taken from Beale et al. (1999) (1); Ercoli et al. (1999) (2); Cadoux et al.

(2008) (3) and Cosentino et al. (2007) (4). The more marked difference in France was due to an interaction with nitrogen supply, where available, errors bars indicate LSD ( $P \leq 0.05$ )

in terms of biomass production and its components. The response of other *Miscanthus* species to water availability is then described under controlled conditions.

Irrigation exerts an important influence on yield when in M.  $\times$  giganteus is grown at sites with a poor water supply (Christian and Haase, 2001). Heaton et al. (2004) described a highly significant effect of water availability on biomass production (P < 0.0001) in a review of several experiments throughout Europe. Under varying levels of nitrogen inputs (between 60 and 240 kg  $N.ha^{-1}$ ), biomass production increased by between 25% and 84% with irrigation (Fig. 2). The difference in yield between well-irrigated plots (100% of evapotranspiration restored)) and less irrigated plots (25% of evapotranspiration restored) was higher in autumn than in winter: +84% for an autumn harvest against 26% for a winter harvest (Fig. 2). Price et al. (2004) found similar losses of harvestable yield in the winter, of between 8% and 23%, by using a potential yield model simulating limited water supplies in England and Wales. These marked differences between rainfed and irrigated yields could be related to

the period during which the drought occurred. Ritchter et al. (2008) determined the main growing season as the period of susceptibility to drought in miscanthus.

In terms of biomass components, water availability does not affect shoot production (Christian and Haase, 2001; Cosentino et al., 2007; Cadoux et al., 2008). However, this lack of effect is probably because shoot production takes place during the period of high soil water availability (at the beginning of the growing period). By contrast, the number of stems is more closely dependent on planting densities than on irrigation rate, *i.e.* water availability.

Plant height is not influenced by water availability at the beginning of the growing period. Nevertheless, a reduction in water availability towards the end of the growing period was found to markedly influence plant height (Christian and Haase, 2001; Cosentino et al., 2007; Cadoux et al., 2008). Irrigated plants were 49% taller than those without irrigation. A similar trend was observed for the leaf area index (Cosentino et al., 2007; Cadoux et al., 2008). Irrigation caused a 77% increase in the leaf area index compared with no irrigation. However, this effect was not reported by Christian and Haase (2001). In *Miscanthus*  $\times$  *giganteus*, Cosentino et al. (2007) also observed a one month difference in the flowering date between irrigated and rainfed treatments.

During studies to compare M.  $\times$  giganteus, M. sinensis and M. sacchariflorus under controlled conditions, a reduction in leaf area was observed in  $M. \times$  giganteus and M. sacchariflorus but not in M. sinensis under water stress (Clifton-Brown and Lewandowski, 2000a; Clifton-Brown et al., 2002). However, the leaf area of M. sinensis was smaller before the water stress was applied. Furthermore, the water needs of this species may be less than in the other species. The authors concluded that M. sinensis might be less sensitive to water stress. In addition, M. sacchariflorus and  $M. \times giganteus$  presented an increase in the senescence of green leaf area, while M. sinensis presented a lack of senescence under all treatments. This experiment suggested that the "stay green" mechanism in M. sinensis may be related to stomatal closure with a low soil moisture content. No data are available on the variability of the "stay green" mechanism observed in species of M. sinensis.

Photosynthetic activity contributes to adapting miscanthus to drought. Weng (1993) showed that both stomatal and non-stomatal photosynthesis factors were affected by water deficit, and suggested that genotypes displaying the highest degree of osmotic adjustment (OA) were the best at maintaining photosynthetic activity under water deficit. Nevertheless, no differences in OA were observed among *M. sinensis*, *M. sacchariflorus* and *M.* × *giganteus* under severe water deficit, although *M. sinensis* was more tolerant to water stress (Clifton-Brown et al., 2002). Leaf conductance was markedly reduced in *M. sinensis*, even under mild water stress, so that a completely green leaf area was maintained throughout the experiment.

# 4.4 Effect of Nitrogen Availability on Biomass and Its Components

Plant growth and development are dependent on nitrogen availability. Nitrogen forms part of living cells and is essential to all protein, enzymatic and metabolic processes involved in synthesis and transfer to energy. The response of biomass production and its components to nitrogen supply has been the subject of more study in  $M. \times giganteus$  species (Ercoli et al., 1999; Christian and Haase, 2001; Cosentino et al., 2007; Danalatos et al., 2007; Christian et al., 2008) than in M. sinensis species (Obara, 1967; Matsumura et al., 1975; Sato et al., 1975 and Hoshimo, 1978 cited by Stewart et al., 2009).

Obara (1967), Matsumura et al. (1975), Sato et al. (1975) and Hoshimo (1978) cited by Stewart et al. (2009) showed that nitrogen is one of the most important nutrients required to increase the biomass production of *M. sinensis* grassland. A N input level of around 100 kg N.ha<sup>-1</sup> can achieve a 50% to 500% higher biomass yield, and affect biomass components. Hoshimo (1978) observed a 16%–33% increase in plant height and leaf size under appropriate nitrogen supply.

By contrast, under non-limiting water conditions, nitrogen fertiliser rates of between 60 and 240 kg N.ha<sup>-1</sup> generally had little or no effect on the biomass yield, stem number, plant height and leaf area index of  $M. \times giganteus$  species (Ercoli et al., 1999; Christian and Haase, 2001; Danalatos et al., 2007; Cadoux et al., 2008; Christian et al., 2008).

During a multi-environment experiment, Lewandowski and Schmidt (2006) observed a trend towards increased biomass yields with a N input level of about 110 kg  $N.ha^{-1}$ , followed by a slight decline. However, these results varied considerably, as the same biomass of about 35 t.ha<sup>-1</sup> could be produced with nitrogen inputs ranging from 90 to 150 kg  $N.ha^{-1}$ . This variability might have been explained by the soil nitrogen supply, but these data were not available.

In a Mediterranean environment but with a water supply, Cosentino et al. (2007) found an autumn biomass yield that was 21% higher (120 kg N.ha<sup>-1</sup>) than without nitrogen during the second year of the crop. This increase even reached 35% for a winter harvest. It could be attributed to an 11% increase in plant height and a 27% increase in stem numbers par square metre.

Plant responses to nitrogen depend on the water availability for biomass production. During the same experiment, Cosentino et al. (2007) determined a marked nitrogen effect when the crop received less water during its third year of growth (Table 3). During an extended period of water stress in July, plant height was 53% higher under nitrogen input, but the difference was smaller (22%) when water was abundant during the second year of growth. The nitrogen

		Water supply	r (mm)	Plant height	Above-ground dry biomass
2 years old (1)	$I_0 N_0$ $I_0 N_2$	272 mm	+244%	+2%	+21%
	$I_2N_0 \\ I_2N_2$	938 mm		+22%	+39%
3 years old (1)	$I_0 N_0$ $I_0 N_2$	228 mm	+93%	+47%	+8 %
	$I_2N_0$ $I_2N_2$	439 mm		+53%	+25%
Means of 4 years (2)	$\begin{array}{c} I_0 N_0 \\ I_0 N_1 \end{array}$	228 mm	+100%		+23%
	$\begin{array}{c} I_2N_0\\ I_2N_1 \end{array}$	439 mm			+41%
Means of 4 years (2)	$I_0 N_0$ $I_0 N_3$	228 mm	+100%		+41%
	$\begin{array}{c} I_2N_0\\ I_2N_3 \end{array}$	439 mm			+100%

**Table 3** Higher increase in plant height or above-ground dry biomass of M. × *giganteus* for plots with water supply (I<sub>2</sub>) and nitrogen supply (N<sub>1</sub>, N<sub>2</sub> or N<sub>3</sub>) than for plots with nitrogen supply only. Data taken from Cosentino et al. (2007) (1) and Ercoli et al. (1999) (2)

I<sub>0</sub>, rainfall; I<sub>2</sub>, Irrigation.

 $N_0$ ,  $N_{soil}$ ;  $N_1$ ,  $N_{soil} + 100$  kg N.ha<sup>-1</sup>;  $N_2$ ,  $N_{soil} + 120$  kg N.ha<sup>-1</sup>;  $N_3$ ,  $N_{soil} + 200$  kg N.ha<sup>-1</sup>.

effect also appears to be dependent on crop age, as a weak effect of nitrogen was only observed in two- or three-year-old crops (Cosentino et al., 2007; Lewandowski and Schmidt, 2006). Contrary to what was expected, no nitrogen effect was detected when the crop was mature and after 14 successive spring harvests (Lewandowski et al., 2000; Danalatos et al., 2007; Christian et al., 2008). No data are available on the effect of successive autumn harvests, and it could be expected that these might increase crop nitrogen requirements; the allocation of nitrogen from leaves and stems to rhizomes occurs during the winter.

According to these results,  $M. \times giganteus$  has low nitrogen needs for growth. For example, Long and Beale (2001) found that 93 kg of N availability per hectare enabled the production of 25 t.ha<sup>-1</sup> of above-ground biomass. Nevertheless, Beale and Long (1997) recommended nitrogen input of between 50 and 90 kg.ha<sup>-1</sup>.year<sup>-1</sup> to enable rhizome development and prevent soil exhaustion.

Moreover, Beale et al. (1996) reported no difference in the photosynthetic potential of M. × giganteus under either 120 kg(N).ha<sup>-1</sup> or no nitrogen applications under temperate field conditions. Weng and Hsu (2001) showed that the photosynthetic capacity of seventeen *Miscanthus* clones collected from lowland areas in Northern and Southern Taiwan and mountainous areas in central Taiwan increased slightly with N applications from 8–16  $\mu$ mol.m<sup>-2</sup>.s<sup>-1</sup> with no nitrogen input to 12–21  $\mu$ mol.m<sup>-2</sup>.s<sup>-1</sup> under 176 kg.ha<sup>-1</sup> nitrogen. However, no correlation with biomass production was calculated during this study.

This low need for and response to nitrogen of  $M. \times$ giganteus crops, even after 14 years of winter harvest, could be attributed to efficient nitrogen storage by the rhizome. By studying <sup>15</sup>N-amended soil, Christian et al. (1997) found that plant nitrogen uptake over 2 years was greater than the amount supplied as fertiliser and arising from soil nitrogen. The plants were thus able to obtain nitrogen from other sources. Eckert et al. (2001) identified one of these as a nitrogen-fixing bacterium, Azospirillium, associated with the roots of  $M. \times giganteus$ . Another study found the same type of bacterium (Clostridia) associated with the roots of *M. sinensis* (Miyamoto et al., 2004) and its stems (Ye et al., 2005). No studies have yet been performed to determine the relationships and exchanges between miscanthus roots and these bacteria, but Ye et al. (2005) showed that the inoculation of this type of bacteria in biomass enhanced the tolerance of salinity. It is therefore possible that exchanges between bacteria and miscanthus exist. Moreover, Funatsu (2006) cited by Stewart et al. (2009) reported that AM fungi (Arbuscular mycorrhiza) associated with the roots of M. sinensis in volcanic and ashy soil might enhance nitrogen assimilation.

#### 5 Conclusions and Prospects

Miscanthus has been identified as a good candidate for biomass production. Peak yields of 49 t DM.ha<sup>-1</sup> have been observed in Europe during an autumn harvest of mature crops with irrigation. We observed smaller maximum yields of between 7 and 26 t DM.ha<sup>-1</sup> for a winter harvest, this difference mainly being due to leaf loss. Because of its C<sub>4</sub> photosynthetic pathway and perennial rhizome,  $M. \times giganteus$  displays a satisfactory combination of radiation-, water- and nitrogen-use efficiencies (see Sect. 2.3) when compared with several other bio-energy crops.

The slight effect of nitrogen input on yield increases under water stress. However, lower levels of nitrogen availability need to be tested in the future because soil nitrogen availability and mineralisation have not been addressed by most studies, suggesting that experiments have probably been carried out under nitrogen excess to demonstrate any effect in miscanthus.

The high potential for above-ground biomass production by  $M. \times giganteus$ , and the efficiencies of this crop, nonetheless varied under different management systems and climates. Two key elements for improvement have been identified by the present review. Firstly, frost tolerance during the winter or early spring seems essential in Northern Europe in order to ensure overwintering and protect young shoots after early emergence. Although  $M. \times giganteus$  has been identified as a C<sub>4</sub> cold-tolerant species by comparison with maize because of its higher PPDK protein content at cold temperatures, we did not observe any winter survival of M.  $\times$  giganteus in Northern Europe (North Sweden and Denmark). Susceptibility to winter frost at temperatures below  $-3.5^{\circ}$ C for the rhizome and  $-8^{\circ}$ C for young shoots of *M*. × giganteus may lead to marked plant losses and lower yields. Secondly, water supply is necessary to ensure a good establishment rate and a high biomass yield. We observed a reduction of up to 84% in above-ground dry matter production under water stress for autumn harvests, and up to 26% for winter harvests. This reduction was mainly due to a 49% decrease in plant height and a 77% decrease in leaf area index. Insofar as a lack of water will become a crucial problem in some areas in the decades to come, it is essential to improve the tolerance of miscanthus to water stress and the water-use efficiency of this crop under diverse levels of water supply.

Because no genetic variability exists in  $M. \times gigan$ teus, and because the development of a single clone of this species throughout Europe might facilitate disease development, M. sinensis must provide the genetic resources for frost and water stress tolerance. Through the production of hybrids, this species displays a good potential for biomass yield under European conditions, as well as tolerance to frost and to limited water supplies. Firstly, its better frost tolerance than  $M. \times giganteus$  has been observed in hybrids between two M. sinensis which have been able to resist temperatures as low as -6°C for rhizomes and -9°C for young shoots. And secondly, M. sinensis is able to maintain its leaf area index under water stress during the first stage of growth, even though it has a similar water-use efficiency.

However, to date it has not been possible to combine a high tolerance of abiotic stress with high levels of biomass production. It is now necessary either to identify *M. sinensis* genotypes that can produce an aboveground biomass yield close to that of *M.* × giganteus under water stress and/or low growth temperatures, or to generate new interspecific hybrids of *M.* × giganteus with higher tolerance thresholds. On a crop scale, the cultivation of sterile species, for example with unpaired ploidy levels, would be useful to prevent uncontrolled invasion. As in other genera, this variation in genome ploidy could lead to variations in species traits, particularly with respect to biomass production, but no studies have as yet been reported.

Acknowledgements The authors acknowledge the Picardie region (project PEL), which supported this work. We thank Prof. A. Gallais for his helpful comments on the manuscript. Thanks also to the anonymous reviewers for their valuable recommendations.

#### References

- Angelini L.G., Ceccarini L., Di Nasso N.N., Bonari E. (2009) Comparison of Arundo donax L. and Miscanthus × giganteus in a long-term field experiment in Central Italy: Analysis of productive characteristics and energy balance, Biomass Bioenerg. 33, 635–643.
- Aspinal D. (1961). The control of tillering in the barley plant I. The pattern of tillering and its relation to nutrient supply, Aust. J. Biol. Sci. 14, 93–505.
- Beale C.V., Long S.P. (1995) Can perennial C<sub>4</sub> grasses attain high efficiencies of radiant energy-conversion in cool climates, Plant Cell Environ. 18, 641–650.

- Beale C.V., Long S.P. (1997) Seasonal dynamics of nutrient accumulation and partitioning in the perennial C-4grasses *Miscanthus* × *Giganteus* and *Spartina Cynosuroides*, Biomass Bioenerg. 12, 419–428.
- Beale C.V., Bint D.A., Long S.P. (1996) Leaf photosynthesis in the C<sub>4</sub>-grass Miscanthus × giganteus, growing in the cool temperate climate of southern England, J. Exp. Bot. 47, 267–273.
- Beale C.V., Morison J.I.L., Long S.P. (1999) Water use efficiency of C<sub>4</sub> perennial grasses in a temperate climate, Agr. Forest Meteorol. 96, 103–115.
- Benbi D.K., Brar J.S. (2009) A 25-year record of carbon sequestration and soil porperties in intensive agriculture, Agron. Sustain. Dev. 29, 257–265.
- Boehmel C., Lewandowski I., Claupein W. (2008) Comparing annual and perennial energy cropping systems with different management intensities, Agr. Syst. 96, 224–236.
- Bullard M.J., Nixon P.M.I., Cheath M. (1997) Quantifying the yield of Miscanthus × giganteus in the UK, Aspects Appl. Biol. 49, 199–206.
- Cadoux S., Vanderdriessche V., Machet J.M., Mary B., Beaudouin N., Lemaire G., Gosse G. (2008) Potential yield and main limiting factors of Miscanthus giganteus in France. Identification of the needs for further research, in: 16th European Biomass Conference and Exhibition, Valence.
- Ceotto E. (2008) Grasslands for bioenergy production. A review, Agron, Sustain. Dev. 28, 47–55.
- Christian D.G., Haase E. (2001) Agronomy of Miscanthus, in: Jones M., Walsh M. (Eds.), Miscanthus for energy and fibre, James and James, London, UK, pp. 21–45.
- Christian D.G., Poulton P.R., Riche A.B., Yates N.E. (1997) The recovery of <sup>15</sup>N–labelled fertilizer applied to *Miscanthus* × *giganteus*, Biomass Bioenerg. 12, 21–24.
- Christian D.G., Poulton P.R., Riche A.B., Yates N.E., Todd A.D. (2006) The recovery over several seasons of 15N-labelled fertilizer applied to Miscanthus × giganteus ranging from 1 to 3 years old, Biomass Bioenerg. 30, 125–133.
- Christian D.G., Riche A.B., Yates N.E. (2008) Growth, yield and mineral content of Miscanthus × giganteus grown as a biofuel for 14 succissive harvests, Ind. Crop. Prod. 28, 320–327.
- Christian D.G., Yates N.E., Riche A.B. (2009) Estimation of ramet production from *Miscanthus* × giganteus rhizome of different ages, Ind. Crop. Prod. 30, 176–178.
- Clifton-Brown J.C., Jones M.B. (1997) The thermal response of leaf extension rate in genotypes of the C<sub>4</sub>-grassMiscanthus: an important factor in determining the potential productivity of different genotypes, J. Exp. Bot. 48, 1573–1581.
- Clifton-Brown J.C., Lewandowski I. (2000a) Water Use Efficiency and Biomass Partitioning of Three Different Miscanthus Genotypes with Limited and Unlimited Water Supply, Ann. Bot. 86, 191–200.
- Clifton-Brown J.C., Lewandowski I. (2000b) Overwintering problems of newly established Miscanthus plantations can be overcome by identifying genotypes with improved rhizome cold tolerance, New Phytol. 148, 287–294.
- Clifton-Brown J.C., Lewandowski I. (2002). Screening *Miscanthus* genotypes in field trials to optimise biomass yield and quality in Southern Germany, Eur. J. Agron. 16, 97–100.
- Clifton-Brown J.C., Lewandowski I., Andersson B., Basch G., Christian D.G., Kjeldsen J.B., Jorgensen U., Mortensen J.V., Riche A.B., Sshwarz K.U., Tatebe K., Teixeira F. (2001a)

Performance of 15 Miscanthus genotypes at five sites in Europe, Agron. J. 93, 1013–1019.

- Clifton-Brown J.C., Lewandowski I., Bangerth F., Jones M.B. (2002) Comparative responses to water stress in stay-green, rapid- and slow senescing genotypes of the biomass crop, Miscanthus, New Phytol. 154, 335–345.
- Clifton-Brown J.C., Long S.P., Jorgensen U. (2001b) Miscanthus productivity, in: Jones M., Walsh M. (Eds.), Miscanthus for energy and fibre, James and James, London, UK, pp. 46–67.
- Clifton-Brown J.C., Neilson B., Lewandowski I., Jones M.B. (2000) The modelled productivity of Miscanthus × giganteus (GREEF et DEU) in Ireland, Ind. Crop. Prod. 12, 191–200.
- Clifton-Brown J.C., Stampfl P.F., Jones M.B. (2004) Miscanthus biomass production for energy in Europe and its potential contribution to decreasing fossil fuel carbon emissions, Global Change Biol. 10, 509–518.
- Cosentino S.L., Patane C., Sanzone E., Copani V., Foti S. (2007) Effects of soil water content and nitrogen supply on the productivity of Miscanthus × giganteus Greef et Deu. in a Mediterranean environment, Ind. Crop. Prod. 25, 75–88.
- Crutzen P.J., Mosier A.R., Smith K.A., Winiwarter W. (2008)  $N_2O$  release from agro-biofuel production negates global warming reduction by replacing fossil fuels, Atmos. Chem. Phys. 8, 389–395.
- Danalatos N.G., Archontoulis S.V., Mitsios I. (2007) Potential growth and biomass productivity of Miscanthus × giganteus as affected by plant density and N-fertilization in central Greece, Biomass Bioenerg. 31, 145–152.
- DEFRA (2007) Planting and growing Miscanthus, Best practice guidelines, For Applicants to DEFRA's Energy Crops Scheme.
- Deuter M. (2000) Breeding approaches to improvement of yield and quality in Miscanthus grown in Europe, EMI Project, Final report, pp. 28–52.
- Eckert B., Weber O.B., Gudrun K., Andras H., Marion S., Anton H. (2001) Azospirillum doebereinerae sp. nov., a nitrogen-fixing bacterium associated with the C<sub>4</sub>- grass Miscanthus, Int. J. Syst. Evol. Microbiol. 51, 17–26.
- Ercoli L., Mariotti M., Masoni A., Bonari E. (1999) Effect of irrigation and nitrogen fertilization on biomass yield and efficiency of energy use in crop production of Miscanthus, Field Crop. Res. 63, 3–11.
- Farage P., Blowers D., Long S.P., Baker N.R. (2006). Low growth temperature modify the efficiency of light use by photosystem II for CO<sub>2</sub> assimilation in leaves of two chilling tolerant C<sub>4</sub> species, Cyperus longus L. and Miscanthus × giganteus, Plant Cell Environ. 29, 720–728.
- Farrell A.D., Clifton-Brown J.C., Lewandowski I., Jones M.B. (2006) Genotypic variation in cold tolerance influences the yield of Miscanthus, Ann. Appl. Biol. 149, 337–345.
- Gillet M. (1980) Les graminées fourragères, Gauthiers-Villars, 306 p.
- Greef J.M., Deuter M. (1993) Syntaxonomy of Miscanthus-X-Giganteus Greef-Et-Deu, Angew. Bot. 67, 87–90.
- Greef J.M., Deuter M., Jung C., Schondelmaier J. (1997) Genetic diversity of European Miscanthus species revealed by AFLP fingerprinting, Genet. Resour. Crop Ev. 44, 185–195.
- Gurtler J. (2007) Biocarburants 2010 : Quelles utilisations des terres en France ? Ed ONIGC.
- Hansen E.M., Christensen B.T., Jensen L.S., Kristensen K. (2004) Carbon sequestration in soil beneath long-term

*Miscanthus* plantations as determined by <sup>13</sup>C abundance, Biomass Bioenerg. 26, 97–105.

- Hastings A., Clifton-Brown J.C., Wattenbach M., Stampfl P., Mitchell C.P., Smith P. (2008) Potential of Miscanthus grasses to provide energy and hence reduce greenhouse gas emissions, Agron. Sustain. Dev. 28, 465–472.
- Heaton E.A., Clifton-Brown J., Voigt T.B., Jones M.B., Long S.P. (2004) Miscanthus for renewable energy generation: European Union experience and projections for Illinois, Mitigation and Adaptation Strategies for Global Change 9, 21–30.
- Heaton E.A., Dohleman, F.G., Long S.P. (2008) Meeting US biofuel goals with less land: the potential of Miscanthus, Global Change Biol. 14, 2000–2014.
- Himken M., Lammel J., Neukirchen D., Czypionka-Krause U., Olfs H.-W. (1997) Cultivation of *Miscanthus* under West European conditions: Seasonal changes in dry matter production, nutrient uptake and remobilization, Plant Soil 189, 117–126.
- Hodkinson T.R., Chase M.W., Renvoize S.A. (2002) Characterization of a genetic resource collection for Miscanthus (Saccharinae, Andropogoneae, Poaceae) using AFLP and ISSR PCR, Ann. Bot. 89, 627–636.

Hoshimo (1978) See reference cited in Stewart et al., 2009.

- Jezowski S. (2008) Yield traits of six clones of Miscanthus in the first 3 years following planting in Poland, Ind. Crop. Prod. 27, 65–68.
- Jorgensen U., Muhs H. (2001) Miscanthus: Breeding and Improvement, in: Jones M., Walsh M. (Eds.), Miscanthus for energy and fibre, James and James, London, UK, pp. 68–85.
- Jorgensen U., Mortensen J., Ohlsson C. (2003a) Light interception and dry matter conversion efficiency of miscanthus genotypes estimated from spectral reflectance measurements, New Phytol. 157, 263–270.
- Jorgensen U., Mortensen J., Kjeldsen J.B., Schwarz K.U. (2003b) Establishment, Developement and yield qualoty of fifteen *Miscanthus* genotypes over three years in Denmark, Acta Agr. Scand. B. S. P. 53, 190–199.
- Kalembasa D., JeŁowski S., Pude R., Malinowska E. (2005) The content of carbon, hydrogen and nitrogen in different development stage of some clones of *Miscanthus*, Pol. J. Soil Sci. 38, 169–177.
- Lewandowski I. (1998) Propagation method as an important factor in the growth and development of *Miscanthus* × giganteus, Ind. Crop. Prod. 8, 229–245.
- Lewandowski I. Clifton-Brown J.C. (2000) European Miscanthus Improvement Project (EMI), FAIR3 CT-96-1392, Final Report, 260 p.
- Lewandowski I., Kauter D. (2003) The influence of nitrogen fertilizer on the yield and combustion quality of whole grain crops for solid fuel use, Ind. Crop. Prod. 17, 103–117.
- Lewandowski I., Kicherer A. (1997) Combustion quality of biomass: Practical relevance and experiments to modify the biomass quality of Miscanthus × giganteus, Eur. J. Agron. 6, 163–177.
- Lewandowski I., Schmidt U. (2006) Nitrogen, energy and land use efficiencies of miscanthus, reed canary grass and triticale as determined by the boundary line approach, Agr. Ecosyst. Environ. 112, 335–346.

- Lewandowski I., Clifton-Brown J.C., Scurlock J.M.O., Huisman W. (2000) Miscanthus: European experience with a novel energy crop, Biomass Bioenerg. 19.
- Lewandowski I., Kicherer A., Vonier P. (1995) Co<sub>2</sub>-Balance for the Cultivation and Combustion of Miscanthus, Biomass Bioenerg. 8, 81–90.
- Lewandowski I., Scurlock J.M.O., Lindvall E., Christou M. (2003) The development and current status of perennial rhizomatous grasses as energy crops in the US and Europe, Biomass Bioenerg. 25, 335–361.
- Lewitt J. (1980) Responses of Plants to Environmental Stresses, Vol. 1, Academic Press, New York, pp. 87–90.
- Linde-Laursen I. (1993) Cytogenetic analysis of Miscanthus 'Giganteus', an interspecific hybrid, Hereditas 119, 297–300.
- Long S.P., Beale C.V. (2001) Resource Capture by *Miscanthus*, in: Jones M., Walsh M. (Eds.), Miscanthus for energy and fibre, James and James, London, UK, pp. 10–20.
- Matsumura et al. (1975) See reference cited in Stewart et al., 2009.
- Miyamoto T., Kawahar, M., Minalisawa K. (2004) Novel Endophytic Nitrogen-fixing Clostridia from grass Miscanthus sinensis as reveled by terminal restriction fragment lenght polymorphism analysis, Appl. Environ. Microbiol. 70, 6580–6586.
- Naidu S.L., Moose S.P., Al-Shoaibi A.K., Raines C.A., Long S.P. (2003) Cold tolerance of C<sub>4</sub> photosynthesis in Miscanthus × giganteus: Adaptation in amounts and sequence of C<sub>4</sub> photosynthetic enzymes, Plant Physiol. 132, 1688–1697.
- Obara (1967) See reference cited in Stewart et al., 2009.
- Plazek A., Dubert F., Marzec K. (2009) Cell membrane permeability and antioxidant activities in the rootstocks of Miscanthus × giganteus as an effect of cold and frost treatment, J. Appl. Bot. Food Quality 82, 158–162.
- Price L., Bullard M.J., Lyons H., Anthony S., Nixon P.M.I. (2004) Identifying the yield potential of Miscanthus × giganteus: an assessment of the spatial and temporal variability of M. × giganteus biomass productivity across England and Wales, Biomass Bioenerg. 26, 3–13.
- Ritchter G.M., Riche A.B., Dailey A.G., Gezan S.A., Powlson D.S. (2008) Is UK biofuel supply from *Miscanthus* waterlimited? Soil Use Manage. 24, 235–245.
- Sage R.F, Kubien D.S. (2007) The temperature response of C<sub>3</sub> and C<sub>4</sub> photosynthesis, Plant Cell Environ. 30, 1086–1106.
- Sato et al. (1975) See reference cited in Stewart et al., 2009.
- Schwarz K.U, Murphy D.P.L., Schnug E. (1994) Studies of the growth and yield of Miscanthus × giganteus in Germany, Aspects Appl. Biol. 40, 533–540.
- Shoji S., Kurebayashi T., Yamada I. (1990) Growth and chemical composition of Japanese pampas grass (*Miscanthus sinensis*) with special reference to the formation of dark-colored andisols in northeastern Japan, Soil Sci. Plant Nutr. 36, 105–120.
- Stewart J.R., Toma Y., Fernandez F.G., Nishiwaki A., Yamada T., Bollero G. (2009) The ecology and agronomy of *Miscanthus sinensis*, a species important to bioenergy crop development, in its native range in Japan: a review, Global Change Biol. 1, 126–153.
- Tayot X., Chartier M., Varlet-Grancher C., Lemaire G. (1995) Potential above-ground dry matter production of miscanthus in north-central France compared to sweet sorghum, in:

Chartier P., Beenackers A., Grassi G. (Eds.), Biomass for energy, environment, agriculture and industry, Elsevier, Oxford, UK, pp. 556–564.

- Thorne G.N. (1962) Survival of tillers and distribution of dry matter between ear and shoot of barley varieties, Ann. Bot. 26, 37–54.
- Vargas L.A., Andersen M.N., Jensen C.R., Jorgensen U. (2002) Estimation of leaf area index, light interception and biomass accumulation of Miscanthus sinensis 'Goliath' from radiation measurements, Biomass Bioenerg. 22, 1–14.
- Vonwuhlisch G., Deuter M., Muhs H.J. (1994) Identification of Miscanthus Varieties by Their Isozymes, J. Agron. Crop Sci. 172, 247–254.
- Walsh M. (1998) Miscanthus handbook. Miscanthus productivity Network AIR-CT92-0294, Hyperion Energy Systems Ltd, Cork, Ireland, p. 225.
- Wang D., Naidu S., Portis A.R., Moose S.P., Long S.P. (2008a) Can the cold tolerance of C<sub>4</sub> photosynthesis in Miscanthus × giganteus relative to Zea mays be explained by differences in activities and thermal properties of Rubisco, J. Exp. Bot. 59, 1779–1787.
- Wang D., Portis A.R., Moose S.P., Long S.P. (2008b) Cool C<sub>4</sub> Photosynthesis: Pyruvate P<sub>i</sub> Dikinase Expression and

Activity Corresponds to the Expectional Cold Tolerance of Carbon Assimilation in Miscanthus  $\times$  giganteus, Plant Physiol. 148, 557–567.

- Wang M., Wu M., Huo H. (2007) Lice- cycle energy and greenhouse gas emission impacts of differents corn ethanol plant types, Environ. Res. Lett. 2, 1–13.
- Weng J.-H. (1993) Photosynthesis of different ecotypes of Miscanthus spp. as affected by water stress, Photosynthetica 29, 43–48.
- Weng J.H., Hsu F.-H. (2001) Gas Exchange and Epidermal Characteristics of Miscanthus Populations in Taiwan Varying with Habitats and Nitrogen Application, Photosynthetica 39, 35–41.
- Weng J.H., Ueng R.G. (1997) Effect of temperature on photosynthesis of Miscanthus clones collected from different elevations, Photosynthetica 34, 307–311.
- Yan W.K., Hunt L.A. (1999). An equation for modelling the temperature response of plant using only the cardinal temperature, Ann. Bot. 84, 607–614.
- Ye B., Saito A., Minamisawa K. (2005) Effect of inoculation with anaerobic nitrogen-fixing consortium on salt tolerance of *Miscanthus sinensis*, Soil Sci. Plant Nutr. 51, 243–249.

# **Changes in Atmospheric Chemistry and Crop Health**

Jürgen Bender and Hans-Joachim Weigel

### Contents

1	Intro	duction		
2	Spatial and Temporal Trends of Atmospheric Changes			
3	Crop	Responses to Atmospheric Trace Gases 491		
	3.1	Direct Effects on Crop Growth and Yield 491		
	3.2	Direct Effects on Crop Quality 493		
	3.3	Interactive Effects of Atmospheric		
		Compounds		
	3.4	Indirect Effects 494		
4	Conc	lusion 495		
Re	ferenc	<b>es</b>		

Abstract The concentrations of atmospheric compounds such as greenhouse gases, heavy metals and trace gas air pollutants have rapidly changed. Many of these compounds interact with agricultural systems and influence crop performance, both directly by affecting growth and quality or indirectly by altering the plant's ability to cope with other abiotic and biotic stresses. Some atmospheric compounds have little or no discernible impact on the environment; others reach levels that exceed thresholds for damage to crops. In this review, we analyse the literature on airborne species that directly impact crop growth and health. In Europe and North America emissions of SO<sub>2</sub>, NOx and heavy metals have declined during the past decades and are currently not considered as a major threat to crops. By contrast, air pollutant emissions have been increasing in rapidly growing regions of Asia, Africa and Latin America. Ozone is the most phytotoxic of the common air pollutants. The widespread distribution of O<sub>3</sub> already presents a risk to crop growth and health in many regions of the world. It is concluded that the continuous increase in background O<sub>3</sub> concentrations will pose a critical threat to future world food security. Interactions with both biotic and abiotic factors must be taken into account in assessing risks of air pollutants in the field. There is evidence that these indirect effects could be more important under certain circumstances than the direct effects of air pollutants on plants. The parallel rapid increase in atmospheric CO<sub>2</sub> concentrations accompanied by climate change has two major implications: (1) a possible benefit to crop growth by direct stimulation of photosynthesis and by mitigation of gaseous air pollutants and water stress; and (2) a threat to crop production due to an enhancement of crop quality losses.

J. Bender (🖂)

Institute of Biodiversity, Johann Heinrich von Thünen-Institute (vTI), Federal Research Institute for Rural Areas, Forestry and Fisheries, Bundesallee 50, 38116 Braunschweig, Germany e-mail: juergen.bender@vti.bund.de

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_22,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Bender and Weigel, Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2010013

**Keywords** Atmospheric change • Carbon dioxide • Crops • Ozone • Pollutant interactions • Product quality•Yield

### 1 Introduction

Anthropogenic activities have significantly changed the composition of the global atmosphere. Especially, the concentrations of several trace gases have undergone significant changes during the past century and continue to change. Plants are important mediators in the exchange of the different gaseous and particulate compounds between the atmosphere and the biosphere (Table 1). The transport of these compounds from the atmosphere into vegetation is by dry and wet deposition of gases, aerosols and sedimenting particles. Many atmospheric constituents can influence crop performance, both directly by affecting growth and quality or indirectly by altering the plant's ability to cope with other abiotic and biotic stresses. In terms of their impact on agricultural ecosystems the atmospheric compounds (for abbreviations see Table 1) can be broadly divided into:

 compounds which act as macro- or micronutrients, e.g., the gases CO<sub>2</sub>, SO<sub>2</sub>, NO, NO<sub>2</sub> and NH<sub>3</sub> and particulate NH<sub>4</sub>, NO<sub>3</sub>-N, SO<sub>4</sub>-S, P, Ca, Fe and Mg, and compounds which may cause adverse or toxic effects, e.g., the gaseous pollutants O<sub>3</sub>, SO<sub>2</sub>, NO<sub>2</sub>, NH<sub>3</sub>, HF, PAN, NMHC and VOC, metals such as Pb, Cd and Hg, or excess nutrient substances, e.g., N, S, Zn and Al, which alter normal patterns of growth and development in ecosystems (Dämmgen and Weigel, 1998).

There are both natural and human-made sources for most of the atmospheric constituents. An air pollutant is usually defined as "a chemical constituent added to the atmosphere through human activities resulting in the elevation of its concentration above a background" (Krupa, 1997).

In this review, we focus on those airborne species whose concentrations have a known trend and are directly interfering with agroecosystems. Ambient air is always composed of mixtures of different species, with the concentrations of individual species or pollutants, respectively, varying in time and with location. For example, a particular air pollutant such as SO<sub>2</sub>, HF or NH<sub>3</sub> can be dominant only in the vicinity of its sources, i.e. those pollutants are primarily of local importance. In comparison, among secondary pollutants, O<sub>3</sub> is of widespread global occurrence and can currently be considered to be the most important air pollutant (Fuhrer, 2009). Among the different environmental factors which determine crop growth, recent and predicted further changes in climate such as increased temperature, altered pattern of rainfall intensity and

 Table. 1 Atmospheric compounds involved in element flux between vegetation and the atmosphere (after Dämmgen and Weigel, 1998)

• H <sub>2</sub> O vapour, CO <sub>2</sub> , CH <sub>4</sub> , N <sub>2</sub> O, NO <sub>2</sub> , O <sub>3</sub> $\rightarrow$	trapping of infrared radiation,
	contribution to the greenhouse effect
• NH <sub>3</sub> , CO, HC $\rightarrow$	effects on reactivity of the atmosphere
• CH <sub>4</sub> CO <sub>2</sub> , SO <sub>2</sub> , NO <sub>2</sub> , NO, NH <sub>3</sub> (gases) $\rightarrow$	involved in nutrient cycling,
NH <sub>4</sub> -/NO <sub>3</sub> -N, SO <sub>4</sub> S, P, Ca, K,	act as macro- and micronutrients
Fe, Mg (particles)	
• $O_3$ , $SO_2$ , $NO_2$ , $HF$ , $H_2O_2$ , $PAN$ , $\rightarrow$	potentially toxic, affecting "normal"
NMHC/VOC (gases), heavy metals	growth and performance of organisms,
(e.g., Pb, Cd, Hg), surplus nutrients	populations and ecosystems
(bioavailable forms of N, S, Zn, Al)	

Abbreviations: Al: aluminum; Ca: calcium; Cd: cadmium; CH<sub>4</sub>: methane; CO: carbon monoxide; CO<sub>2</sub>: carbon dioxide; Fe: iron; HC: hydrocarbons; HF: fluoride; Hg: mercury; H<sub>2</sub>O: water vapour; H<sub>2</sub>O<sub>2</sub>: hydrogen peroxide; H<sub>2</sub>S: hydrogen sulphide; K: potassium; Mg: magnesium; N: nitrogen; N<sub>2</sub>O: nitrous oxide; NH<sub>3</sub>: ammonia; NH<sub>4</sub><sup>+</sup>: ammonium; NO: nitrogen monoxide; NO<sub>2</sub>: nitrogen dioxide; NO<sub>3</sub><sup>-</sup>: nitrate; NO<sub>X</sub>: NO + NO<sub>2</sub>; NMHC: non-methane hydrocarbons; O<sub>3</sub>: ozone; P: phosphorus; PAN: peroxyacetyl-nitrate; Pb: lead; S: sulphur; SO<sub>2</sub>: sulphur dioxide; SO<sub>4</sub><sup>2-</sup>: sulphate; VOC: volatile organic compounds; Zn: zinc.

frequency, and atmospheric  $CO_2$  concentration as well as other atmospheric compounds have become and will be of growing importance in many parts of the world. Therefore, projections of future global food security must equally consider the likely impacts of climate change and air pollution. With respect to effects, responses of crops to  $O_3$  and  $CO_2$  are particularly considered here as these trace gases are key variables of climatic and atmospheric change for future global food production (Long et al., 2005; Vandermeiren et al., 2009).

# 2 Spatial and Temporal Trends of Atmospheric Changes

The concentrations of SO<sub>2</sub>, nitrogen oxides (NOx) and VOCs as well as of heavy metals and photochemical oxidants such as O<sub>3</sub> in many parts of the industrialised world have changed significantly during the last century (Dämmgen and Weigel, 1998). While local emissions of urban or industrial sources still occur, emissions, particularly of SO<sub>2</sub> and to a smaller extent of NOx (NO + NO<sub>2</sub>), VOCs and particulate matter, have declined during the past decades in Europe and North America. This was due to successful policies to reduce emissions, as well as a decline in polluting heavy industries (UNECE, 2007). For example, emissions of SO<sub>2</sub> throughout Western Europe have declined by approximately 80% since the peak in emissions during the 1970s (Fowler et al., 2001). SO<sub>2</sub> levels and sulphur bulk deposition are now usually low during the growth periods of crops. At present, annual mean SO2 concentrations in Europe are less than 10  $\mu$ g m<sup>-3</sup> and rarely exceed 50  $\mu$ g m<sup>-3</sup> SO<sub>2</sub>, and bulk S depositions are lower than 10-15 kg ha<sup>-1</sup>  $a^{-1}$  (Fowler et al., 2001). Emission estimates of atmospheric nitrogen species (NOx, NH<sub>3</sub>) are more uncertain because of the variety of gases and sources (Grübler, 2003). However, the concentrations in oxidised atmospheric N compounds also show a declining trend although NOx concentrations are highly variable due to different local traffic densities (UNECE, 2007). In rural regions of Europe annual mean concentrations for NO<sub>2</sub> range between 5 and 30  $\mu$ g m<sup>-3</sup> and less than 5  $\mu$ g m<sup>-3</sup> for NO (Dämmgen and Weigel 1998). Currently, the emissions of NH<sub>3</sub> from farm land determine the concentrations of this gas in the air above these systems. The concentrations range from 1 to 30 µg NH<sub>3</sub>, if one excludes periods with applications of liquid manure or slurry. Concentrations of airborne VOCs are also highly diverse and the evaluation of their occurrence and distribution is difficult, because there are both anthropogenic and biogenic sources, the latter including emissions from plant, animal and microbial sources (Kesselmeier and Staudt, 1999; Cape, 2003; Krupa et al., 2008). Average annual concentrations of the major airborne VOCs benzene, toluene and ethylene are usually lower than 5  $\mu$ g m<sup>-3</sup>, although maximum hourly concentrations of some VOCs can be 100 times larger than the average (Collins and Bell, 2002; Cape, 2003). For the majority of heavy metals such as Pb, Cd, Ni, Hg and Zn a similar decline in emission and subsequent deposition has been observed since the late 1980s in most of Europe, although higher metal deposition is still found in some Eastern European countries (Harmens et al., 2008).

In contrast to the situation in Europe and North America, air pollutant emissions have been increasing over the last two decades in many developing countries, particularly in rapidly growing regions of Asia, Africa and Latin America, where rapid industrialisation and population growth is taking place accompanied by increasing energy demand and road traffic, but with poor emission controls (Emberson et al., 2003). China and India are now the leading emitters of  $SO_2$  in the world (Marshall, 2002). Also, the predicted increase in global NOx emissions may be attributed largely to the high percentage increases in developing countries, such as China (Marshall, 2002; Ghude et al., 2009).

Nitrogen oxides and VOCs are important precursors for the formation of tropospheric  $O_3$ . Tropospheric  $O_3$  is a widespread secondary air pollutant found in all industrialised countries worldwide, and meanwhile also in many of the developing countries in the world, where it has reached levels in ambient air which are of concern with respect to vegetation damage and human health effects (Emberson et al., 2003; Royal Society, 2008), and these trends are expected to continue as economies continue to expand. While at least in most parts of Western Europe there is a clear trend of decreasing  $O_3$  peak values ("photosmog episodes"), predictive models indicate that background  $O_3$  concentrations will continue to increase at a rate of 0.5% to 2% per year in the Northern Hemisphere during the



**Fig. 1** Predicted changes in global surface  $O_3$  concentration from 2000 to 2100. The figure shows the averaged changes in  $O_3$  concentrations (ppb) for January and July. Blue colours de-

note no or little change while green to red colours point to a medium to severe increase in  $O_3$  concentrations. Adapted from Prather et al. (2003)

next several decades. Currently, the background  $O_3$  concentration in the Northern hemisphere is within the range of 23–34 ppb; however, global surface  $O_3$  concentration is expected to be in the range of 42–84 ppb by 2100 (Vingarzan, 2004). Figure 1 shows the projected global increase in  $O_3$  concentration over the next 100 years from Prather et al. (2003), based on IPCC global emission scenarios. According to these,

the locations of the major  $O_3$  increases ("hot-spots") in the future are expected to be Asia, Southern Africa, Southern Europe and the USA.

In contrast to the different temporal trends of the "classical" air pollutants such as  $SO_2$  and NOx between industrialised and developing countries and the respective spatial variability in their concentration levels, atmospheric  $CO_2$  concentration has risen steadily all over the globe from a pre-industrial concentration of about 280 ppm to a current value of about 385 ppm, and could reach more than 550 ppm by 2050 (IPCC, 2007). Due to the direct effects of rising  $CO_2$  levels on crop photosynthesis, growth and quality, assessments of future air pollution effects on plants and crops have to consider this rapid change.

Overall, the concentrations of sulphur and nitrogenbased air pollutants and of heavy metals have declined during the past decades in many countries of Europe and North America. By contrast, air pollutant emissions have been increasing in rapidly growing regions of Asia, Africa and Latin America. Tropospheric  $O_3$ , however, is still at high levels worldwide.

### 3 Crop Responses to Atmospheric Trace Gases

## 3.1 Direct Effects on Crop Growth and Yield

Gaseous atmospheric compounds are transferred from the atmosphere onto plant canopies by diffusion, which is governed by micro-meteorological conditions (radiation, temperature, wind, etc.). The major path of entry into the leaf is through the stomata. The other two pathways are non-stomatal pathways, which include deposition on external plant surfaces including the cuticle and deposition on the soil (Fig. 2). Penetration of gases through plant cuticles is usually of minor importance (Lendzian and Kerstiens, 1991), although some pollutants such as  $SO_2$  are able to damage the cuticle and gain entry to the internal leaf tissue to some extent (Wellburn, 1994). Aerosols and sedimenting particles containing nutrients and pollutants (e.g., heavy metals) are deposited on plant surfaces or on soil surfaces directly; matter deposited on plant surfaces indirectly can be transferred to the soil by run-off or by plant debris or litter. After entering the root zone they are available for plant uptake. The reaction of a plant to a given air pollutant depends on the exposure characteristics, plant properties and external growth conditions (Krupa, 1997; Bender and Weigel, 2003). Short-term exposures to relatively high concentrations generally result in acute visible foliar injury. Long-term chronic exposures to lower concentrations can cause alterations in physiological and biochemical processes that may result in chlorosis, premature senescence, and growth and yield reductions.

For example, while exposure to short-term high concentrations of  $SO_2$  can lead to cellular death, exposure to low to moderate  $SO_2$  concentrations over the long term can result in impaired cellular metabolism, an accumulation of sulphate in the vacuole, and a reduction in photosynthetic rate (Legge et al., 1998). Sensitivity of plants to  $SO_2$  is generally determined by mesophyll resistance and by buffering capacities



Fig. 2 Major pathways of the transfer of air pollutants to terrestrial surfaces (plant canopies and soil). The thickness of the arrows denotes the relative importance of the respective pathway. Redrawn by courtesy of Dr. Lisa Emberson, SEI York, UK

(i.e. the ability to counteract a change in pH from an optimal value; Wellburn, 1994). However, current ambient SO<sub>2</sub> concentrations in Europe can no longer be considered as a serious problem with respect to negative effects on plant performance. The few data describing exposure-response relationships for longterm (weeks to months) exposure to low and moderate SO<sub>2</sub> concentrations (50 to 100  $\mu$ g m<sup>-3</sup>) on crop growth and yield are somewhat variable and controversial (Weigel et al., 1990; Legge and Krupa, 2002). The overall evidence from fumigation studies shows that for most annual crop species ambient SO<sub>2</sub> concentrations are below the critical levels above which adverse effects occur (UNECE, 2004). The reduction in SO<sub>2</sub> emissions in Europe and North America has even resulted in the occurrence of sulphur deficiency in some agricultural species (e.g., rape) growing in sulphur-deficient areas (Legge and Krupa, 2002).

Similarly, phytotoxic effects of reactive nitrogen species (NO, NO<sub>2</sub>, NH<sub>3</sub>), particularly on crops, can only be observed at concentrations which are far above those occurring even in heavily polluted environments (Wellburn, 1994; UNECE, 2004). Exposure to moderate levels of NO<sub>2</sub> can stimulate certain metabolic processes (e.g., enzyme activities), which indicates that airborne nitrogen derived from NO2 can be incorporated into the plant's nitrogen metabolism (Wellburn, 1990; Bender et al., 1991). At current rural concentrations, NO<sub>2</sub> is unlikely to be phytotoxic but may act, to some extent, as an additional source of N (Davison and Cape, 2003). Also, the concentrations of  $NH_3$ in rural areas cannot be considered as a problem for crop growth, although there is some evidence that high and low temperatures or drought stress can considerably modify the effects of NH<sub>3</sub> (Cape et al., 2009). Overall, in highly fertilised agricultural systems across Europe atmospheric nitrogen (NO<sub>2</sub>/NO, NH<sub>3</sub>) and sulphur (SO<sub>2</sub>, H<sub>2</sub>S) compounds at current ambient levels cannot be considered as a direct threat for annual crops. By contrast, the few research findings from experiments performed in some developing countries of South and East Asia so far suggest that these pollutants can already lead to serious reductions of crop growth and yield, a situation which may be exacerbated in the future (Marshall, 2002; Ishii et al., 2004; Fu et al., 2007).

Considering current concentrations of VOCs, especially in rural areas, direct impacts of most of these compounds on crops are unlikely or unknown. Experiments to study the direct effects of VOCs on plants are usually performed in the laboratory and have used very high concentrations relative to ambient air (Cape, 2003). While a wide range of tolerance to airborne VOCs among plant species has been demonstrated, a few experiments have shown specific effects, particularly of ethylene (a plant hormone), on reproductive stages (seed germination, flowering, fruit ripening) (Collins and Bell, 2002; Cape, 2003).

For agriculture, chronic effects of air pollutants such as  $O_3$  are of particular concern, because they are due to exposures for weeks, months, or over the entire lifecycle of the crop. It is well known that increasing  $O_3$  levels cause a decline in the yield of many crop species, such as wheat, rice, soybean and cotton (Ashmore, 2005). Such yield losses have been attributed to reduced photosynthetic rate, altered carbon allocation and accelerated leaf senescence (Fiscus et al., 2005; Fuhrer, 2009). Recently, Mills et al. (2007) analysed O<sub>3</sub> exposure-response data for 19 agricultural and horticultural crops, respectively, and identified wheat, water melon, pulses, cotton, turnip, tomato, onion, soybean and lettuce as the most ozone-sensitive crops, while, for instance, barley was classified as O<sub>3</sub>resistant. Holland et al. (2006) estimated crop losses and the associated economic loss in Europe for 23 horticultural and agricultural crops for the base year 2000 and found an overall loss of 3% of all crop species considered, which would be equivalent to  $\in$  6.7 billion economic damage. The global impact of  $O_3$  on crop yields was recently evaluated by Van Dingenen et al. (2009). Their estimates of present day global relative yield losses ranged between 7% and 12% for wheat, between 6% and 16% for soybean, between 3% and 4% for rice, and between 3% and 5% for maize. When translating the production losses into global economic damage for the four crops considered, they estimated an economic loss in the range of \$14-26 billion. About 40% of this damage is occurring in China and India. However, the uncertainty on these estimates is large. This is primarily due to the O<sub>3</sub> exposure metrics used in the estimates, which are based on the exposure concentrations in ambient air, either on a regional, national or global scale, rather than on the actual uptake of O<sub>3</sub> and thus do not account for the dose-specific nature of plant responses. In addition, only the direct O<sub>3</sub> effects on crop growth are considered, while indirect growth effects, e.g., mediated by phytosanitary problems, are not taken into account (see Sect. 3.3). Moreover, wide

variability in  $O_3$  sensitivity among various cultivars of a crop is common (USEPA, 2006). In summary, tropospheric  $O_3$  remains the most important atmospheric pollutant that has direct negative effects on many crop species worldwide.

By contrast, the continuous rise in atmospheric  $CO_2$ levels will principally have a positive effect on crop growth and yield, as CO<sub>2</sub> directly affects plant physiology and growth by serving as a primary substrate for photosynthesis. Generally, elevated CO2 concentrations can increase biomass and yield in C<sub>3</sub> crops by increasing photosynthesis and decreasing photorespiration, but with large differences among species in the magnitude of the yield stimulation (Amthor, 2001; Kimball et al., 2002; Ainsworth and Long, 2004). No significant stimulation of yield has been found so far in C<sub>4</sub> crops, at least under well-watered conditions, because C<sub>4</sub> photosynthesis is saturated under ambient  $CO_2$  (Long et al., 2005). However, in all crops (both  $C_3$ ) and  $C_4$ ), higher  $CO_2$  concentrations reduce stomatal conductance and transpiration and improve water-use efficiency, i.e. crops will experience a reduced demand for water.

as is the case for a number of horticultural crops. For example, Kostka-Rick et al. (2002) have shown that environmentally-relevant concentrations of  $O_3$  can cause visible foliar injury on species such as lettuce, spinach or onion, which would make these crops unmarketable (Figs. 3 and 4).



**Fig. 3** Ozone injury on spinach (*Spinacea oleracea* cv. Matador). Courtesy: J. Bender, vTI Braunschweig, Germany

#### 3.2 Direct Effects on Crop Quality

In comparison to air pollutant and climate change effects on crop growth and yield, much less is known about potential effects on the quality or the nutritive value, respectively, of agricultural and horticultural crops. Changes in crop quality due to O<sub>3</sub> exposure have been studied in a limited number of crops. For example, in wheat, O<sub>3</sub> reduced yield but increased grain protein concentration (Pleijel et al., 1999; Piikki et al., 2008). Moreover, O<sub>3</sub> was found to have positive effects on the quality of potato tubers by decreasing reducing sugars and increasing the vitamin C content (Vorne et al., 2002). In contrast,  $O_3$  has been found to reduce the oil, protein and carbohydrate contents of the seeds of rape (Ollerenshaw et al., 1999). Recent evidence suggests that O<sub>3</sub> can also alter the plant food quality for ruminant animals. Decreased nutritive quality of forages was found in a number of pasture species (Krupa et al., 2004; Bender et al., 2006).

Pollutant-induced visible injury is of particular significance when the quality and the marketable value of the crop depend on the appearance of the foliage,



**Fig. 4** Ozone injury on spring onion (*Allium fistulosum* cv. Polo). Courtesy: J. Bender, vTI Braunschweig, Germany

A frequently observed phenomenon is that plants grown at high CO<sub>2</sub> levels exhibit significant changes in their chemical composition (Idso and Idso, 2001; Loladze, 2002). A prominent example of a CO<sub>2</sub> effect is the decrease in the nitrogen (N) concentration in vegetative plant parts as well as in seeds and grains and, related to this, the decrease in the protein concentrations (Cotrufo et al., 1998; Taub et al., 2008; Wieser et al., 2008). Other CO<sub>2</sub> enrichment studies have shown changes in the composition of other macro- and microelements (Ca, K, Mg, Fe, Zn) and in concentrations of secondary compounds, vitamins and sugars (Idso and Idso, 2001). Overall, these CO<sub>2</sub>induced changes may have negative consequences with respect to nutritional quality of foods and feeds, the plant-herbivore interaction and the element turnover of ecosystems, respectively. The examples above indicate that there may be economically important effects of air pollution and climate changes on the quality of crops and forage species, although the available information is still inconsistent.

# 3.3 Interactive Effects of Atmospheric Compounds

Under field conditions plants are exposed to different environmental factors including more than only one atmospheric compound. Based primarily on experimental work it has been shown that mixtures of atmospheric compounds and air pollutants, respectively, modify the magnitude and nature of the response to individual compounds. Generally, pollutant combinations may result in either more-than-additive (synergistic) or less-than-additive (antagonistic) effects. Based on the prevailing conditions at that time interactions of O<sub>3</sub> with other air pollutants (e.g., SO<sub>2</sub>, NO<sub>2</sub>) were studied quite frequently in the 1980s (reviewed by Fangmeier et al., 2002). Currently, at least for Europe and North America, a simultaneous occurrence of O<sub>3</sub>, SO<sub>2</sub>, NO<sub>2</sub> or NH<sub>3</sub> at phytotoxic levels is rather unusual and far less frequent than sequential or combined sequential/concurrent exposures. From experiments where exposure conditions have been more realistic in terms of their likelihood of occurrence in ambient air it can be concluded that: (1) antagonistic interactions tend to be found when gases were applied sequentially (e.g.,  $O_3/NO_2$ ) and/or when, e.g., nitrogenous or sulphurous air pollutants were combined with  $O_3$  at relatively low levels, suggesting that plants were able to utilise the additional S or N source; and (2) synergistic interactions are more likely to be found when  $O_3$  was applied simultaneously with another pollutant at high concentrations (Fangmeier and Bender, 2002). For the situation in Europe and North America this would imply that neither SO<sub>2</sub> nor NO<sub>2</sub> seems likely to pose an additional risk to that related to  $O_3$ . However, the effects of pollutant combinations on crop growth and yield should have a much higher significance in many developing countries where air pollutants such as SO<sub>2</sub>, NOx and  $O_3$  are rapidly increasing (Ishii et al., 2007).

With respect to the future there is some evidence that elevated  $CO_2$  has the potential to mitigate negative effects of O<sub>3</sub>(and other gaseous pollutants), mainly due to a CO<sub>2</sub>-induced reduction in stomatal conductance, which reduces O<sub>3</sub> uptake. On the other hand, O<sub>3</sub> limits positive CO<sub>2</sub> responses in many plants as well (Fiscus et al., 2005). All climate change factors, such as CO<sub>2</sub>, warming, changes in precipitation, etc., which may affect stomatal conductance and thus the flux of gaseous air compounds into leaves, will exert a modification on the effects of individual pollutants (Bender and Weigel, 2003; Harmens et al., 2007). In summary, the available information suggests that the continuing increases in the CO<sub>2</sub> component of climate change are likely to be ameliorative for the effects of air pollutants. Although there is some evidence that the concomitant increases in mean global temperature may decrease these protective effects (Fuhrer, 2003; USEPA, 2006), it is still not possible to predict the combined impacts of climate change and air pollutants on crops.

# 3.4 Indirect Effects

Atmospheric compounds and air pollutants, respectively, may interact with other biotic and abiotic growth or stress factors (e.g., water and nutrient supply; heat and water stress; salinity, pesticide application; pests and pathogens; symbiotic relationships) in a complex manner, thus causing indirect effects on crop performance. For example, while it is well accepted that reduced vitality due to  $O_3$  stress can make plants more susceptible to plant pathogens, general predictions of O<sub>3</sub> effects on particular plant-pathogen systems are difficult, because the available data for specific pests and diseases are often controversial (USEPA, 2006; Fuhrer, 2009). Increased susceptibility after  $O_3$  exposure has been reported for necrotrophic pathogens, while obligate biotrophic infections tend to be diminished by O<sub>3</sub> (Manning and von Tiedemann, 1995; USEPA, 2006). With regard to insect pathogens, there is a general trend that some pests may have a preference for and grow better when feeding on O<sub>3</sub>-stressed plants, but there are also other observations where insect growth was not changed (USEPA, 2006). Viral infection often provides some protection from O<sub>3</sub> injury; however, the type and degree of protection depend on the specific host and virus (Manning and von Tiedemann, 1995).

The direct effects of elevated CO<sub>2</sub> levels on tissue chemical composition can have an indirect effect on plant-herbivore interactions, as host plants growing under enriched CO<sub>2</sub> environments usually exhibit, e.g., decreased tissue N concentration, increased C/N ratios and generally altered secondary metabolism of C-based secondary and structural compounds. This in turn may affect food consumption by herbivores and related population development (Stiling and Cornelissen, 2007). However, there is almost no information about how O<sub>3</sub> effects on plant-pathogen systems may be modified in a future climate with elevated CO<sub>2</sub> (Chakraborty et al., 2000; Fuhrer, 2009). For example, while host plants growing under enriched CO<sub>2</sub> environments usually exhibit larger biomass, increased C/N ratios and decreased tissue N concentration, O<sub>3</sub> has the opposite effect (Pleijel et al., 1999; Piikki et al., 2008). Hence, it remains open how food consumption by herbivores and population development are affected under future atmospheric conditions characterised by elevated O<sub>3</sub> and CO<sub>2</sub> concentrations (Stiling and Cornelissen, 2007).

Another important interaction may occur between the effects of air pollutants and soil moisture availability. Water supply directly affects stomatal conductance and hence the uptake and effects of gaseous air pollutants. For example, it is known that reduced soil moisture limits  $O_3$  uptake by decreasing stomatal conductance, which increases  $O_3$  tolerance (Bender and Weigel, 2003). However, other findings suggest that, in some species, soil moisture stress may reduce rather than increase  $O_3$  tolerance (Bungener et al., 1999). The complex physiological and morphological changes due to water deficit impair plant vitality itself, e.g., by promoting senescence processes. Therefore, decreased pollutant uptake may not necessarily be connected with decreased pollutant injury. As outlined before (Sect. 3.1), elevated  $CO_2$  concentrations often improve water-use efficiency, i.e. may mitigate drought stress effects (Manderscheid and Weigel, 2007), which is an important feedback effect in future climate change scenarios.

Although the available information is clearly insufficient to understand the importance of interactions between air pollutants and biotic or abiotic factors, it is suggested that these indirect effects could be more important under certain circumstances than the direct effects of the gases on plants.

### 4 Conclusion

Crops, similarly to all other types of vegetation, are closely linked to the exchange of matter between the atmosphere and biosphere. After deposition of atmospheric compounds on canopies, crop growth and quality may be affected in various ways. Regarding the situation in most parts of Europe and North America, exposure to compounds such as SO2, NO2/NO, VOCs and heavy metals is reduced and is currently not a major threat to crops. However, in many regions of both continents continuously increasing background levels of tropospheric  $O_3$  remain a problem, which poses an additional risk to crop growth and health during the growing season. In the growing economies of many developing countries the concentrations of atmospheric compounds such as SO<sub>2</sub>, NOx, NH<sub>3</sub> and particularly O<sub>3</sub> are rapidly increasing. Already, these pollutants can lead to serious reductions of crop growth and yields, a situation which may be exacerbated in the future. Interactions with both biotic and abiotic factors must be taken into account in assessing risks of air pollutants in the field. On a global scale the rapid change in atmospheric composition by the increase in the atmospheric CO<sub>2</sub> concentration accompanied by climate change has two major implications. A possible benefit to crop growth by direct stimulation of photosynthesis and by mitigation of gaseous air pollutants and water stress, but concomitantly a threat to crop production due to an enhancement of crop quality losses.

#### References

- Ainsworth E.A., Long S.P. (2005) What have we learned from 15 years of free-air CO2 enrichment (FACE)? A metaanalytic of the responses of photosynthesis, canopy properties and plant production to rising CO<sub>2</sub>, New Phytol. 165, 351–372.
- Amthor J.S. (2001) Effects of atmospheric CO<sub>2</sub> on wheat yield: review of results from experiments using various approaches to control CO<sub>2</sub> concentrations, Field Crops Res. 73, 1–34.
- Ashmore M.R. (2005) Assessing the future global impacts of ozone on vegetation, Plant Cell Environ. 28, 949–964.
- Bender J., Weigel H.-J. (2003) Ozone stress impacts on plant life, in: Ambasht R.S., Ambasht N.K. (Eds.), Modern Trends in Applied Terrestrial Ecology, Kluwer Academic/Plenum Publishers, New York, pp. 165–182.
- Bender J., Muntifering R.B., Lin J.C., Weigel H.J. (2006) Growth and nutritive quality of *Poa pratensis* as influenced by ozone and competition, Environ. Pollut. 142, 109–115.
- Bender J., Weigel H.J., Jäger H.J. (1991) Response of nitrogen metabolism in beans (*Phaseolus vulgaris* L.) after exposure to ozone and nitrogen dioxide, alone and in sequence, New Phytol. 119, 261–267.
- Bungener P., Nussbaum S., Grub A., Fuhrer J. (1999) Growth response of grassland species to ozone in relation to soil moisture conditions and plant strategy, New Phytol. 142, 283–294.
- Cape J.N. (2003) Effects of airborne volatile organic compounds on plants, Environ. Pollut. 122, 145–157.
- Cape J.N., van der Eerden L.J., Sheppard L.j., Leith I.D., Sutton M.A. (2009) Evidence for changing the critical level for ammonia, Environ. Pollut. 157, 1033–1037.
- Chakraborty S., von Tiedemann A., Teng P.S. (2000) Climate change: potential impact on plant diseases, Environ. Pollut. 108, 317–326.
- Collins C.D., Bell J.N.B. (2002) Effects of volatile organic compounds, in: Bell J.N.B., Treshow M. (Eds.), Air pollution and plant life, John Wiley & Sons, Chichester, pp. 173–186.
- Cotrufo M.F., Ineson P., Scott A. (1998) Elevated CO<sub>2</sub> reduces the nitrogen concentration of plant tissue, Glob. Change Biol. 4, 43–54.
- Dämmgen U., Weigel H.J. (1998) Trends in atmospheric composition (nutrients and pollutants) and their interaction with agroecosystems, in: El Bassam N., Behl R., Prochnow B. (Eds.), Sustainable agriculture for food, energy and industry: strategies towards achievement, James & James Science Publishers Ltd., London, pp. 85–93.
- Davison A.W., Cape J.N. (2003) Atmospheric nitrogen compounds – issues related to agricultural systems, Environ. Int. 29, 181–187.
- Emberson L.D., Ashmore M.R., Murray F. (2003) Air Pollution Impacts on Crops and Forests: a Global Assessment, Imperial College Press, London.
- Fangmeier A., Bender J. (2002) Air pollutant combinations: significance for future impact assessments on vegetation, Phyton 42, 65–71.
- Fangmeier A., Bender J., Weigel H.J., Jäger H.J. (2002) Effects of pollutant mixtures, in: Bell J.N.B., Treshow M. (Eds.), Air pollution and plant life, John Wiley & Sons, Chichester, pp. 251–272.

- Fiscus E.L., Booker F.L., Burkey K.O. (2005) Crop responses to ozone: uptake, modes of action, carbon assimilation and partitioning, Plant Cell Environ. 28, 997–1011.
- Fowler D., Sutton M.A., Flechard C., Cape J.N., Storeton-West R., Coyle M., Smith R.I. (2001) The control of SO<sub>2</sub> dry deposition on natural surfaces and its effects on regional deposition, Water Air Soil Pollut.: Focus 1, 39–48.
- Fu B.J., Zhuang X.L., Jiang G.B., Shi J.B., Lü Y.H. (2007) Environmental problems and challenges in China, Environ. Sci. Technol. 41, 7597–7602.
- Fuhrer J. (2003) Agroecosystem responses to combinations of elevated CO<sub>2</sub>, ozone, and global climate change, Agric. Ecosyst. Environ. 97, 1–20.
- Fuhrer J. (2009) Ozone risk for crops and pastures in present and future climates, Naturwissenschaften 96, 173–194.
- Ghude S.D., van der A.R.J., Beig G., Fadnavis S., Polade S.D. (2009) Satellite derived trends in NO<sub>2</sub> over the major global hotspot regions during the past decade and their intercomparison, Environ. Pollut. 157, 1873–1878.
- Grübler A. (2003) Trends in global emissions: Carbon, sulfur, and nitrogen, in: Douglas I. (Ed.), Encyclopedia of global environmental change. Vol 3: Causes and consequences of global environmental change, John Wiley & Sons, Chichester, pp. 35–53.
- Harmens H., Mills G., Emberson L.D., Ashmore M.R. (2007) Implications of climate change for the stomatal flux of ozone: A case study for winter wheat, Environ. Pollut. 146, 763–770.
- Harmens H., Norris D. and the participants of the moss survey (2008) Spatial and temporal trends in heavy metal accumulation in mosses in Europe (1990–2005), Programme Coordination Centre for the ICP Vegetation, Bangor, UK.
- Holland M., Kinghorn S., Emberson L., Cinderby S., Ashmore M., Mills G., Harmens H. (2006) Development of a framework for probabilistic assessment of the economic losses caused by ozone damage to crops in Europe, CEH Project No. C02309NEW, Report to U.K, Department of Environment, Food and Rural affairs.
- Idso S.B., Idso K.E. (2001) Effects of atmospheric CO<sub>2</sub> enrichment on plant constituents related to animal and human health, Environ. Exp. Bot. 45, 179–199.
- IPCC (Intergovernmental Panel on Climate Change) (2007) The 4th Assessment Report. Working Group I Report: The physical scientific basis. www.ipcc.ch.
- Ishii S., Bell J.N.B., Marshall F.M., (2007) Phytotoxic risk assessment of ambient air pollution on agricultural crops in Selangor State, Malaysia, Environ. Pollut. 150, 267–279.
- Ishii S., Marshall F.M., Bell J.N.B., Abdullah A.M. (2004) Impact of ambient air pollution on locally grown rice cultivars (*Oryza sativa* L.) in Malaysia, Water Air Soil Pollut. 154, 187–201.
- Kesselmeier J., Staudt M. (1999) Biogenic volatile organic compounds (VOC): an overview on emission, physiology and ecology, J. Atmos. Chem. 33, 23–88.
- Kimball B.A., Kobayashi K., Bindi M. (2002) Responses aof agricultural crops to free-air CO<sub>2</sub>-enrichment, Adv. Agron. 77, 293–368.
- Kostka-Rick R., Bender J., Bergmann E., Weigel H.J. (2002) Symptoms of ozone-induced foliar injury on horticultural crops, in: Klumpp A., Fomin A., Klumpp G., Ansel W.

(Eds.), Bioindication and air quality in European cities, G. Heimbach, Stuttgart, pp. 191–196.

- Krupa S., Booker F., Bowersox V., Lehmann C.T., Grantz D. (2008) Uncertainties in the current knowledge of some atmospheric trace gases associated with US agriculture: A review, J. Air Waste Manage. Assoc. 58, 986–993.
- Krupa S., Muntifering R., Chappelka A. (2004) Effects of ozone on plant nutritive quality characteristics for ruminant animals, The Botanica 54, 1–12.
- Krupa S.V. (1997) Air pollution, people, and plants, APS Press, St. Paul, MN, USA.
- Legge A.H., Jäger H.J., Krupa S.V. (1998) Sulfur dioxide, in: Flagler R.B. (Ed.), Recognition of air pollution injury to vegetation: A pictorial atlas, Air Waste Manage. Assoc., Pittsburgh, Pennsylvania, USA, Chap. 3, pp. 1–43.
- Legge A.H., Krupa S.V. (2002) Effects of sulphur dioxide, in: Bell J.N.B., Treshow M. (Eds.), Air pollution and plant life, John Wiley & Sons, Chichester, pp. 135–162.
- Lendzian K.J., Kerstiens G. (1991) Sorption and transport of gases and vapors in plant cuticles, Reviews Environ. Contam. Toxicol. 121, 65–128.
- Loladze I. (2002) Rising atmospheric CO<sub>2</sub> and human nutrition: toward globally imbalanced plant stoichiometry, Trends Ecol. Evol. 17, 457–461.
- Long S.P., Ainsworth E.A., Leakey A.D.B., Morgan P.B. (2005) Global food insecurity. Treatment of major food crops with elevated carbon dioxide or ozone under large-scale fully open.air conditions suggests recent models may have overestimated future yields, Philos. Trans. R. Soc. B 360, 2011–2020.
- Manderscheid R., Weigel H.J. (2007) Drought stress effects on wheat are mitigated by atmospheric CO<sub>2</sub> enrichment, Agron. Sustain. Dev. 27, 79–87.
- Manning W.J., v. Tiedemann A. (1995) Climate change: Potential effects of increased atmospheric carbon dioxide (CO<sub>2</sub>), ozone (O<sub>3</sub>), and ultraviolet-B (UV-B) radiation on plant diseases, Environ. Pollut. 88, 219–245.
- Marshall F.M. (2002) Effects of air pollutants in developing countries, in: Bell J.N.B., Treshow M. (Eds.), Air pollution and plant life, John Wiley & Sons, Chichester, pp. 407–416.
- Mills G., Buse A., Gimeno B., Holland M., Emberson L., Pleijel H. (2007) A synthesis of AOT40-based response functions and critical levels of ozone for agricultural and horticultural crops, Atmos. Environ. 41, 2630–2643.
- Ollerenshaw J.H., Lyons T., Barnes J.D. (1999) Impacts of ozone on the growth and yield of field grown winter oilseed rape, Environ. Pollut. 104, 53–59.
- Piikki K., De Temmerman L., Ojanperä K., Danielsson H., Pleijel H. (2008) The grain quality of spring wheat (*Triticum aestivum* L.) in relation to elevated ozone uptake and carbon dioxide exposure, Eur. J. Agron. 28, 245–254.
- Pleijel H., Mortensen L., Fuhrer J., Ojanpera K., Danielsson H. (1999) Grain protein accumulation in relation to grain yield of spring wheat (*Triticum aestivum* L.) grown in opentop chambers with different concentrations of ozone, carbon dioxide and water availability, Agric. Ecosyst. Environ. 72, 265–270.

- Prather M., Gauss M., Berntsen T., Isaksen I., Sundet J., Bey I., Brasseur G., Dentener F., Derwent R., Stevenson D., Grenfell L., Hauglustaine D. Horowitz L., Jacob D., Mickley L., Lawrence M., von Kuhlmann R., Muller J., Pitari G., Rogers H., Johnson M., Pyle J., Law K., van Weele M., Wild O. (2003) Fresh air in the 21st century, Geophys. Res. Lett. 30, 1–4.
- Royal Society (2008) Ground-level ozone in the 21st century: future trends, impacts and policy implications, Science Policy Report 15/08, The Royal Society, London.
- Stiling P., Cornelissen T. (2007) How does elevated carbon dioxide (CO<sub>2</sub>) affect plant-herbivore interactions? A field experiment and meta-analysis of CO<sub>2</sub>-mediated changes on plant chemistry and herbivore performance, Glob. Change Biol. 13, 1823–1842.
- Taub D.R., Miller B., Allen H. (2007) Effects of elevated CO<sub>2</sub> on protein concentration of food crops: a meta-analysis, Glob. Change Biol. 14, 1–11.
- UNECE (United Nations Economic Commission for Europe) (2004) Manual on methodologies and criteria for Modelling and Mapping Critical Loads and Levels an Air Pollution Effects, Risks and Trends, www.icpmapping.org.
- UNECE (United Nations Economic Commission for Europe) (2007) Review of the Gothenburg Protocol. Report of the Task Force on Integrated Assessment Modelling and the Centre for Integrated Assessment Modelling, CIAM report 1/2007, Bilthoven, Laxenburg.
- USEPA (US Environmental Protection Agency) (2006) Air Quality Criteria for Ozone and Related Photochemical Oxidants, Report no. EPA/600/R-05/004aF-cF, US Environmental Protection Agency, Washington, DC.
- Vandermeiren K., Harmens H., Mills G., De Temmerman L. (2009) Impacts of ground level ozone on crop production in a changing climate, in: Singh S.N. (Ed.), Climate Change and Crops, Environmental Science and Engineering, Springer, Berlin, pp. 213–243.
- Van Dingenen R., Dentener F.J., Raes F., Krol M.C., Emberson L., Cofala J. (2009) The global impact of ozone on agricultural crop yields under current and future air quality legislation, Atmos. Environ. 43, 604–618.
- Vingarzan R. (2004) A review of surface ozone background levels and trends, Atmos. Environ. 38, 3431–3442.
- Vorne V., Ojanpera K., DeTemmerman L., Bindi M., Högy P., Jones M.B., Lawson T., Persson K. (2002) Effects of elevated carbon dioxide and ozone on tuber quality in the European multiple-site experiment CHIP-projekt, Eur. J. Agron. 17, 369–381.
- Weigel H.J., Adaros G., Jäger H.J. (1990) Yield responses of different crop species to long-term fumigation with sulfur dioxide in open-top chambers, Environ. Pollut. 67, 15–28.
- Wellburn A.R. (1990) Why are atmospheric oxides of nitrogen usually phytotoxic and not alternative fertilizers? New Phytol. 115, 395–429.
- Wellburn A.R. (1994) Air pollution and climate change, Longman Scientific and Technical, New York.
- Wieser H., Manderscheid R., Erbs M., Weigel H.J. (2008) Effects of elevated CO<sub>2</sub> concentrations on the quantitative protein composition of wheat grain, J. Agric. Food Chem. 56, 6531–6535.
# Modelling Soil Carbon and Nitrogen Cycles During Land Use Change

J. Batlle-Aguilar, A. Brovelli, A. Porporato, and D.A. Barry

## Contents

1 Introduction				
	1.1	Land Use Change: Forest Versus Agricultural		
		Soils	500	
	1.2	Modelling of Soil C and N Cycles	501	
2	C an	d N Cycles in Soil	502	
3	Mois	sture Dynamics as a Controlling Factor of Soil		
	Cart	oon and Nitrogen Cycles	504	
4	Exis	ting Models of Soil Carbon and Nitrogen		
	Turn	lover	506	
5	Desc	ription of the C-N Model	509	
6	Mod	elling Scenarios	512	
	6.1	Climate Scenarios	512	
	6.2	Soil Scenarios	513	
7	Resu	llts	515	
8	Con	clusions	519	
Re	eferen	ces	522	

D.A. Barry (🖂)

Abstract Forested soils are being increasingly transformed to agricultural fields in response to growing demands for food crop. This modification of the land use is known to result in deterioration of soil properties, in particular its fertility. To reduce the impact of the human activities and mitigate their effects on the soil, it is important to understand the factors responsible for the modification of soil properties. In this paper we reviewed the principal processes affecting soil quality during land use changes, focusing in particular on the effect of soil moisture dynamics on soil carbon (C) and nitrogen (N) cycles. Both physical and biological processes, including degradation of litter and humus, and soil moisture evolution at the diurnal and seasonal time scales were considered, highlighting the impact of hydroclimatic variability on nutrient turnover along with the consequences of land use changes from forest to agricultural soil and vice-versa. In order to identify to what extent different models are suitable for long-term predictions of soil turnover, and to understand whether some simulators are more suited to specific environmental conditions or ecosystems, we enumerated the principal features of the most popular existing models dealing with C and N turnover. Among these models, we considered in detail a mechanistic compartment-based model. To show the capabilities of the model and to demonstrate how it can be used as a predictive tool to forecast the effects of land use changes on C and N dynamics, four different scenarios were studied, intertwining two different climate conditions (with and without seasonality) with two contrasting soils having physical properties that are representative of forest and agricultural soils. The model incorporates synthetic time series of stochastic precipitation, and therefore soil moisture evolution through

Institute for Environmental Engineering, École Polytechnique Fédérale de Lausanne, Station 2, 1015, Lausanne, Switzerland e-mail: andrew.barry@epfl.ch

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_23,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Batlle-Aguilar et al., Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2010007

time. Our main findings in simulating these scenarios are that (1) forest soils have higher concentrations of C and N than agricultural soils as a result of higher litter decomposition; (2) high frequency changes in water saturations under seasonal climate scenarios are commensurate with C and N concentrations in agricultural soils; and (3) due to their different physical properties, forest soils attenuate the seasonal climate-induced frequency changes in water saturation, with accompanying changes in C and N concentrations. The model was shown to be capable of simulating the long term effects of modified physical properties of agricultural soils, being thus a promising tool to predict future consequences of practices affecting sustainable agriculture, such as tillage (leading to erosion), ploughing, harvesting, irrigation and fertilization, leading to C and N turnover changes and in consequence, in terms of agriculture production.

**Keywords** Soil organic matter • Biogeochemical cycles • Agricultural soil • Forest soil • Soil nutrients • Soil moisture dynamics • Soil restoration

#### 1 Introduction

Soils are complex systems sustaining life on Earth. Among other functions, soils maintain plant and animal growth, recycle nutrients and organic wastes, filter and purify water. Precisely, soil quality refers to a combination of chemical, physical, and biological processes that confers to the soil the ability to carry out, among others, these particular ecological functions. Numerous human activities however utilise soil, modify its physical and chemical properties and change the composition of its ecosystems. As a result, in the last century a widespread decrease of soil quality has been observed, together with a deterioration of its functioning (Brady and Weil, 2004).

The main component of soils is organic matter (SOM), which shows a variable degree of decomposition, from fresh litter to highly decomposed humus. SOM stores three to four times the amount of carbon (C) than found in all living vegetation. Other than C, soils also contain nearly all the macro- (nitrogen, N, phosphorous, P, and potassium, K) and micro-nutrients required by living organisms. Among the

macronutrients N plays a major role since it is essential for life but its bio-available forms are seldom abundant in the environment. Therefore in many ecosystems the N cycle controls the overall soil turnover and functioning. For these reasons and without neglecting the importance of other nutrients, in this paper the focus is on soil C and N cycles.

# 1.1 Land Use Change: Forest Versus Agricultural Soils

The increasing demand on food crops, pasture, firewood and timber is at the origin of worldwide changes of land-use in forested areas. This situation is worrying in some areas of the planet, such as South America, where 12% and 7% of forestland was converted to pasture and croplands, respectively, between 1850 and 1985 (Houghton et al., 1991). Land-use changes, and especially cultivation of previously forested land, reduce significantly the soil quality (e.g., changes in SOM content and decomposition rates, changes in soil chemical and physical properties), leading to a permanent degradation of land productivity (Nye and Greenland, 1964; Islam et al., 1999). Furthermore, it has been reported that deforestation increases carbon dioxide (CO<sub>2</sub>) release to the atmosphere (Houghton, 2002), which contributes to global warming.

All studies that focused on the effects of land conversion from forest to cultivated land concluded that land-use change induces a reduction of the available soil C and a decrease in its quality. The maximum rate of loss occurs during the first 10 y of cultivation, with total C decrease up to 30% (Davidson and Ackerman, 1993; Lugo and Brown, 1993; Murty et al., 2002) followed by reduced but still significant reduction rate (Brams, 1971; Martins et al., 1991; Bonde et al., 1992; Motavalli et al., 2000). Furthermore, it was reported that the loss rate is highly variable and influenced by several factors such as the native vegetation, climate, soil type and management practices (Mann, 1986; Davidson and Ackerman, 1993; Bruce et al., 1999).

Contrasting with the conversion from forest to cultivated land, controversy exists when the change is from forest to pasture lands. The overall change in soil C has been shown to be either positive or negative. For instance, de Moraes et al. (1996) found an increase up to 20% in total soil C 20 y after the change in land use, while Veldkamp (1994) reported a net soil organic C loss up to 18% after 25 y. Johnson (1992) also observed that changes in soil C in both land-use cultivation and pasture were associated with changes in soil N. Reiners et al. (1994) found that the transformation of forest land to pasture led to important changes in the N cycling. For example, the ammonium  $(NH_4^+)$  pool was larger in pasture lands while the nitrate  $(NO_3^-)$  pool was less important in pasture than forest lands. This is consistent with a low rate of plant uptake of  $NH_4^+$  and slow nitrification rates (Vitousek, 1984; Vitousek and Sanford, 1986).

One of the important aspects that affect SOM cycling in the transition from forest to cultivated soil is the removal of most of the fresh organic C (litter) due to harvesting (Smil, 1999). However, harvesting is not the only factor responsible for the soil organic C loss. Some other processes that were also recognized to contribute to change the amount of soil C are the changes in litter chemical properties (Feigl et al., 1995; Ellert and Gregorich, 1996; Scholes et al., 1997), soil type (Feller and Beare, 1997; Scholes et al., 1997; García-Oliva et al., 1999), microbial community (Prasad et al., 1995), changes in soil N cycling (Dalal and Mayeer, 1986; Brown and Lugo, 1990; Desjardins et al., 1994) and management practices (Feller and Beare, 1997; Fernandes et al., 1997; Bruce et al., 1999). Soil tillage and ploughing promote redistribution of residues and their decomposition. As a result, soil C and N pools are depleted and soil fertility is lost. Soil C is oxidized to CO<sub>2</sub> and lost to the atmosphere contributing to the increase of greenhouse gases in the atmosphere. Moreover, tillage improves soil aeration, destroys macro-aggregates and changes the hydrological cycle, with an increase of the respiration rates and ultimately an additional depletion of the C pool (Juo and Lal, 1979; Agboola, 1981; Ellert and Gregorich, 1996; Reicosky et al., 1997; Bruce et al., 1999).

In agricultural areas, the root zone (soil depth affected by plant roots) remains constant over time and is relatively shallow. Different rooting patterns have direct effects on the C flux, since they affect soil porosity and soil aeration (Berger et al., 2002). Therefore, changes in land use resulting in a modified rooting depth often have a direct influence on soil respiration and C mineralization rates, and thus on soil turnover (Howard and Howard, 1993). Ecological restoration is the process of assisting the recovery of an ecosystem that has been degraded, damaged or destroyed as a consequence of human activities (Young et al., 2005), and typically involves a land use change. During the restoration, environmental conditions (e.g., type of vegetation, ecosystem corridors or soil practices) are manipulated to create ecological conditions suitable for the successful establishment of a target composition of species (Prober et al., 2005). The change from agricultural soil to the original forest is a typical example of soil restoration, where natural soil properties and vegetation are amended, resulting in an improvement of soil fertility and an ecosystem close to its former natural condition.

#### 1.2 Modelling of Soil C and N Cycles

Numerical tools are becoming increasingly used to understand the modifications induced in ecosystems as a result of changes in land use, and it has been found that understanding the coupled N and C dynamics is of primary importance for predictive models of SOM evolution, for example to changes in land use and responses to global changes (Rodriguez-Iturbe et al., 2001). Modelling of soil biogeochemical processes dates back to the 1930s (Manzoni and Porporato, 2009), and nowadays an extended list of stochastic, empirical and mechanistic models incorporating soil nutrient dynamics is available. Models vary significantly in terms of complexity and mathematical description of the biological and geochemical processes involved. Manzoni and Porporato (2009) reviewed and classified about 250 different mathematical models developed over 80 y. Most of the models currently available evolved from early efforts to provide a concise mathematical description of the soil cycles, and have been adapted and improved to specific applications. The aims of the different models are numerous and include, for example: understanding and prediction of feedbacks between terrestrial ecosystems and global climate (e.g., estimate and predict climatological and biological effects of human activities) (Agren et al., 1991; Melillo, 1996; Moore et al., 2005); influence of climate changes on nutrient cycling in soils (Pastor and Post, 1986; Hunt et al., 1991; Moorhead et al., 1999; Eckersten et al., 2001; Ito, 2007); prediction of changes in soil C and N cycles related to possible land use changes (Eckersten and Beier, 1998; Paul and Polglase, 2004; Christiansen et al., 2006; Findeling et al., 2007; Pansu et al., 2007; Post et al., 2007; Kaonga and Coleman, 2008); and forecasts of crop productivity and system response under specific physical soil changes (Wolf et al., 1989; Wolf and Van Keulen, 1989; Matus and Rodríguez, 1994; Parton and Rassmussen, 1994; Henriksen and Breland, 1999; Nicolardot et al., 2001).

The aim of this manuscript is to provide an overview of the main processes, mechanisms and parameters affecting the evolution of selected soil nutrient cycles (soil C and N) and to provide a modelling framework that incorporates the key mechanisms. Both physical and biological processes, including degradation of litter and humus, and soil moisture evolution on diurnal and seasonal time scales are considered. In the first part of the manuscript, soil C and N cycles are summarized, followed by an overview of the most popular models dealing with soil nutrient turnover. In the second part, a compartment model based on Porporato et al. (2003) is described and applied to simulate soil C and N dynamics, as well as degradation and transformation processes occurring under different precipitation and soil scenarios. Contrasting soil types and precipitation regimes are considered, to illustrate modelling capabilities and to show how numerical tools can be used to understand effects of land use changes over soil C and N fluxes and, thus, the feasibility and viability of ecological restoration regarding the modelled ecosystem and surroundings.

#### 2 C and N Cycles in Soil

The global C cycle can be depicted as consisting of a series of interconnected compartments (terrestrial, aquatic and atmospheric) where C is stored and transformed. Soils are part of the terrestrial C pool (Fig. 1). The amount of C stored in the (living and dead) organic matter in soils is three to four times higher than that in the atmosphere (Bruce et al., 1999). The circulation rates are also high. For these reasons, soil C turnover is of primary importance to developing understanding and forecasting global changes in biogeochemical cycles and climate change (Stevenson and Cole, 1999; Rodriguez-Iturbe and Porporato, 2004). The total global emission of  $CO_2$  from soils is probably the largest flux in the global C cycle, and small changes in the magnitude of soil respiration, if they take place at large scale, could have a tremendous effect on the concentration of  $CO_2$  in the atmosphere (Schlesinger and Andrews, 2000; Murty et al., 2002). By the same argument, soils have also a great potential for long-term C storage. Whether a soil will act as a sink or source of CO<sub>2</sub> depends on a number of environmental factors, including climatic variability and anthropogenic changes in land use, which for example may result in a modified composition of the vegetation and therefore of the quality and quantity of litter inputs (Gignoux et al., 2001).

The principal C exchange processes between soil and atmosphere are *photosynthesis* and *respiration*. Photosynthetic C fixation by plants – often named



Fig. 1 Simplified representation of the global carbon cycle

primary producers – converts atmospheric  $CO_2$  and is the main source of soil organic C. Briefly, during photosynthesis  $CO_2$  is used as a C source to produce complex organic molecules, using sunlight as an energy source (e.g., Killham and Foster, 1994):

$$CO_2 + H_2O + Energy \rightarrow CH_2O + O_2.$$
 (1)

The complex organic molecules produced by plants enter the soil C cycle as decaying organic matter (litter) and are progressively converted to simpler molecules. A significant fraction of the organic C introduced in the soil is directly used as an energy source to sustain pedofauna metabolism, and is released again to the atmosphere in form of  $CO_2$  through respiration:

$$CH_2O + O_2 \rightarrow CO_2 + H_2O + Energy.$$
 (2)

Another part of the soil C is assimilated by vegetation and finally transferred to the soil as plant litter, becoming part of SOM (Porporato et al., 2003). Organic C is available in soils in a large variety of forms. Killham and Foster (1994) partitioned the soil organic C into three main pools: insoluble, soluble and biomass. Insoluble soil organic C includes plant residues and partially decomposed material, which forms the litter and the humus. Soluble C is a fraction of the humus further decomposed and is rapidly assimilated as a substrate by the pedofauna. The fast consumption of soluble C explains its often low concentration in the soil (1%) in comparison to insoluble organic C (90%). Soil biomass (9%) consists of microbes and animals (e.g., macroinvertebrates), the decomposition activity of which is mostly responsible for the C decomposition and recycling (Killham and Foster, 1994).

Within the soil, organic C is transferred between the different pools (or compartments) by means of *decomposition* processes, which are regulated by environmental conditions (e.g., soil moisture) and the C/N ratio (Brady and Weil, 2004). These factors will be discussed subsequently. Litter undergoing decomposition is mainly composed of plant residues (fallen leaves, roots, etc.). Decomposition rates are highly variable in time, and are mainly controlled by the environmental conditions (e.g., soil moisture level, aeration, soil temperature) and the quality of the added litter. Complex organic molecules can be decomposed under either aerobic or anaerobic conditions. Under normal conditions, soils are unsaturated and thus  $O_2$  is likely



to be always available. However, even in the vadose zone saturated conditions can result from, for example, significant precipitation events. Wetlands are a particular case where saturated conditions are found permanently or seasonally. In general, microbial decomposition rates are larger under aerobic conditions (Brady and Weil, 2004), where  $O_2$  acts as the electron acceptor during oxidation of organic compounds (Barry et al., 2002). On the other hand, slow decomposition rates under anaerobic conditions can result in accumulation of considerable amounts of partially decomposed organic matter (Fig. 2).

Soil N comes mainly from the atmosphere, which is the largest N pool and contains almost 75% of the total N available on Earth (Barbour et al., 1999). In brief, the soil N cycle is based upon the uptake of the inorganic forms ( $NO_3^-$ ,  $NH_4^+$ ) by plants. N returns to the soil in organic form as plant residues, which are decomposed by the soil pedofauna (e.g., invertebrates, microbes, fungi) and are made available to plants in inorganic form.

The total amount of organic N in soils varies greatly and is influenced by the soil-forming factors likely climate, topography, vegetation, parent material and age. The N cycle is tightly coupled to the C cycle, since most of the microbial N transformations (e.g., nitrification) use energy supplied by C (Paul, 1976). Although locally N is also incorporated into soils through dry or wet direct deposition, the largest fraction of soil organic N fixation is done biologically (conversion



from  $N_2$  gas to organic forms, mediated by specific microbial strains). N is found in soils mainly within the organic matter fraction, for example humic compounds, plant roots, microbial biomass and decomposing organic materials. The amount of organic N contained in soils far exceeds that which is present in plant-available inorganic forms.

The soil N, mainly present in organic form as previously mentioned, is almost unavailable for plants. The vegetation mainly uses inorganic forms of N, which are made available by the SOM decomposition. Soil microorganisms convert the N contained in the organic matter to  $NH_4^+$  in a process named *mineralization* (Schinner et al., 1995), further subdivided into two processes. The organic N is initially transformed via *ammonification*, and – if O<sub>2</sub> is available –  $NH_4^+$  is subsequently oxidized to nitrite ( $NO_2^-$ ) and  $NO_3^-$ , through *nitrification*:

$$NH_4^+ + O_2 + H^+ + 2e^- \rightarrow NH_2OH + H_2O$$
  
 $\rightarrow NO_2^- + 5H^+ + 4e^-, (3)$ 

$$NO_2^- + H_2O \to NO_3^- + 2H^+ + 2e^-.$$
 (4)

Although plants can use both forms of inorganic N,  $NO_3^-$  is used in preference to  $NH_4^+$  because of its greater solubility in water. In other words, nitrates quickly dissolve in the pore solution, which is taken up by plants. On the other hand, however, this also means that  $NO_3^-$  is easily flushed to groundwater.  $NH_4^+$  is instead less mobile because it is strongly adsorbed on clay minerals due to its positive charge.

*Denitrification* is the anaerobic microbial reduction of N, and  $NO_3^-$  is used as an electron acceptor (i.e., source of energy), resulting in a transfer of soil N to the atmosphere (Groffman et al., 2002):

$$2NO_3^- + 10e^- + 12H^+ \to N_2(g) + 6H_2O.$$
 (5)

*Immobilisation* is a process involving microbial uptake of nutrients, where inorganic N is converted into organic form, such as amino acids and biological macromolecules.

The carbon-to-nitrogen ratio (C/N) is an important factor affecting the overall turnover rates of SOM (Young and Young, 2001). Bacterial sensitivity to the C/N ratio is due to the fact that bacteria need a constant C/N ratio, while this ratio is highly variable in substrate. For example, intense competition among microorganisms for available N occurs when soil residues have a high C/N ratio, i.e., the substrate is poor in N making it the limiting factor (Brady and Weil, 2004). Environmental conditions (e.g., soil moisture and temperature) have a direct influence on bacterial activity and thus on this ratio (Koch et al., 2007). The C/N ratio of the substrate tends to decrease as the SOM becomes more decomposed - from fresh litter to highly transformed humus - when microbes are solely responsible for decomposition (Zheng et al., 1999) because the microbial C/N ratio is lower than that of litter (Persson, 1983). In other words, the humus is enriched in N compared to the litter. For this reason the C/N ratio of the litter pool controls the rates of mineralization/immobilisation. Young and Young (2001) identified a threshold of the C/N ratio which determines the bacterial activity. When C/N > 25, microbes respire completely using the available C and thus assimilate the entire N mineralized, and consequently N is immobilised. In contrast, if C/N < 25, the SOM N content far exceeds the immobilisation capacity of microbial populations and the result is a net mineralization. Although this threshold seems to be directly related to the C/N ratio needs of bacteria, White (1997) argued that this threshold value is variable among different ecosystems, for example because the C/N ratio of the vegetation changes depending on the composition and relative frequency of each species. For example, pines produce litter with C/N ratio as high as 90, while litter originating from cereal crops has C/N ratio of 80 and tropical forest trees produce litter with C/N ratios around 30 (Young and Young, 2001).

# 3 Moisture Dynamics as a Controlling Factor of Soil Carbon and Nitrogen Cycles

Soil moisture results from the interactions between climate, soil type (texture, granulometry, organic matter content) and vegetation, and it is consequently variable both in space and time. Among the possible physical processes the dynamics of soil moisture exerts the greatest influence over SOM turnover, mineralization, decomposition, leaching and uptake, and its effects are complex and non-linear. As an example to illustrate this complexity, the production of plant residues - the main source of litter and therefore of energy for the pedofauna – depends on the growth rate of vegetation, which is controlled by water availability. Accumulation of SOM can increase the water retention capacity of the soil, with a positive feed-back on the vegetation. Moreover, the soil biota activity depends on the soil water content, and optimal decomposition rates are only achieved within a relatively narrow soil moisture range.

Soil biota is sensitive to moisture level for several reasons. In order to preserve cell integrity, when the soil water content decreases bacteria increase the intracellular solute concentration to compensate for the extracellular concentration and counterbalance the increased osmotic pressure (Stark and Firestone, 1995; Bell et al., 2008). Therefore, a high concentration of solutes results in an inhibition of the enzymatic activity and therefore decreased cellular activity. Additionally, as the soil becomes drier, water in soil pores becomes a thin layer covering soil grains and substrate availability becomes diffusion-limited. In consequence, microbial activity is further reduced (Csonka, 1989; Stark and Firestone, 1995; Fenchel et al., 1998).

It is however difficult to identify a unique threshold moisture level under which soil respiration (or microbial activity) diminishes. Davidson et al. (1998) and Rey et al. (2002) estimated that 75% of the soil field capacity corresponds to the soil moisture level below which soil respiration decreases, while according to Xu et al. (2004) a more likely value is 42%.

A number of studies have shown that soil moisture effects on soil C and N turnover also depend on the time-scale of interest. Curiel Yuste et al. (2007) found that, at the seasonal scale, the effect of temperature and soil moisture on CO<sub>2</sub> efflux (e.g., soil respiration) was very similar for ponderosa pine and oak savannah ecosystems. For shorter time scales (e.g., daily), decomposition of organic matter was mainly controlled by temperature during wet periods and a combination of temperature and soil moisture during dry periods. Soil bacterial growth (or soil respiration) - a parameter often used as a measure of microbial activity – shows a maximum at about 30°C (Pietikåinen et al., 2005). Nevertheless, the influence of temperature on microbial activity is generally considered much less important than soil moisture (Rodriguez-Iturbe and Porporato, 2004) because, although important differences in soil temperature are likely to occur at a daily and seasonal scale in the uppermost soil (e.g., first few centimetres), yearly average values at depth are much more constant than those of soil moisture.

Typically, summer drought decreases substantially decomposition rates (Curiel Yuste et al., 2007), but it has been observed that sporadic rains during these dry periods tends to increase the decomposition efficiency of the bacterial communities (Borken et al. 1999, 2002; Savage and Davidson, 2001; Goulden et al., 2004; Xu et al., 2004; Misson et al., 2005; Scott-Denton et al., 2006). Kieft et al. (1998) and Moore et al. (2008) observed an increase of root density and soil microbial activity rate in response to isolated moisture pulses in arid soils, although the response of root density occurred at longer time-scale. A fast rewetting of the soil profile is likely to have negative consequences on microbial populations in that it can generate an osmotic shock and result in cell lyses (Kieft et al., 1987; Van Gestel et al., 1993). In contrast, Ryel et al. (2004), Schwinning and Sala (2004) and Bell et al. (2008) found that, in arid soils, while plants usually do not take advantage of brief pulses of moisture generated by short precipitation events, microbial mineralization is stimulated. Consequently, short-term increases in soil microbial activity triggered by moisture pulses will not typically correlate with an increase in primary production at the same time scale, confirming that plant growth is not only dependent on soil microbial activity, but also on other factors such as the precipitation event duration, amount of soil water infiltrated and the overall change in soil moisture. The magnitude and timing of intra-seasonal precipitation becomes therefore a key regulator for microbial activity (Bell et al., 2008). Since decomposition and consequent mineralization can be stimulated by moisture pulses that are too brief to benefit primary producers (e.g., plants) (Cui and Caldwell, 1997; Schwinning et al., 2003; Austin et al., 2004), in arid soils there is potential for soil nutrient pools to accumulate over time and become available to plants as heavier precipitation occurs. The influence of soil moisture over soil nutrient dynamics has been also studied in temperate (Davidson et al., 1998; Buchmann, 2000; Reichstein et al., 2003) and tropical forests (Conant et al., 2000; Davidson et al., 2000; Kiese and Butterbach-Bahl, 2002; Epron et al., 2004). It was concluded that a strong influence of the soil moisture over microbial activity exists, but that the degree of correlation varies strongly among different ecosystems (Buchmann, 2000; Rustad et al., 2000).

# 4 Existing Models of Soil Carbon and Nitrogen Turnover

At least 250 models dealing with soil C and nutrient turnover exist (Manzoni and Porporato, 2009). Classification of all these simulators is difficult because they are based on a wide range of physical and biogeochemical descriptions of the processes and the underlying assumptions vary significantly. Nevertheless, based on their internal structure models describing SOM dynamics can be divided into (1) process-oriented, (multi)-compartment models; (2) organism-oriented (food-web) models; (3) cohort models describing decomposition as a continuum; and (4) a combination of model types (1) and (2) (Brussaard, 1998; Smith et al., 1998; Post et al., 2007). Process-oriented or compartment models (each compartment or pool is a fraction of SOM with similar chemical and physical characteristics) are built considering the processes involved in the migration of SOM across the soil profile and its transformations (Smith et al., 1998). Models belonging to this class can potentially have a variable degree of complexity, from the simplest case with no compartments (considering degradation as a continuum) to more refined, multi-compartment models, with each compartment composed of organic matter with similar chemical composition or degradability. Processoriented models can be combined with GIS software, giving a modelling platform well suited for regionalscale studies. Examples of successful coupling between soil turnover and GIS software are CANDY (Franko, 1996), CENTURY (Schimel et al., 1994) and RothC (Post et al., 1982; Jenkinson et al., 1991). On the other hand, the theoretical compartments that define the structure of multi-compartment process-oriented models are difficult to compare with the measurements of SOM fractions, and therefore the testing and validation is difficult and limited (Christensen, 1996; Elliott et al., 1996). Among the most popular process-oriented models are also DAISY (Hansen et al., 1991), NC-SOIL (Molina et al., 1983) and SOILN (Johnson et al. 1987).

In organism-based models the SOM flows from one organism pool to another, which in turn are classified depending upon their taxonomy or metabolism. The main advantage of organism-oriented models is that the main drivers of SOM fluxes and transformations – the pedofauna – are explicitly accounted for. However, as noted in Post et al. (2007), to date there is no general acceptance of the existence of a relation between soil biota abundance and degradation rates. In contrast, the relationship between degradation rate and amount (or concentration) of substrate, as in process-based models, is well recognized. Simple first-order kinetic rates are often suitable to model the transformations, and the reaction rates can be easily estimated from laboratory experiments and directly used in process-oriented models. Site-specific calibration of organism-oriented models involves the characterization of the soil microbial consortia and therefore requires more complex techniques, while process-oriented models are less influenced by the features of the microbial communities, and have a larger range of application to different environments. To summarize, process-oriented models are easier to apply and calibrate than organism-oriented, which explains their greater popularity. Nevertheless, organism-oriented models have been proposed by several authors, including Moore et al. (2004), Kuijper et al. (2005), Zelenev et al. (2006) and Cherif and Loreau (2009).

A cohort is a set of items sharing some particular characteristic. Cohort models divide SOM into cohorts, which are further divided into different pools (e.g., C, N). Contrary to process-based models where decay is usually treated as a purely physical or biochemical process, e.g., described by a first-order rate, cohort models consider explicitly microbial physiology as the driving factor of decomposition. An example of a cohort model was proposed by Furniss et al. (1982), where SOM was divided into three cohorts considering age, origin and size, with each cohort subdivided into a number of chemical constituents. Gignoux et al. (2001) developed SOMCO (soil organic matter cohort), where SOM is divided into different cohorts in a demographic sense, meaning that a cohort is a set of items of the same age. At each time step a new cohort is defined and its fate is followed until its relative amount to total SOM becomes negligible. Other examples of models belonging to this class are those of Pastor and Post (1986), Bosatta and Ågren (1991, 1994) and Frolking et al. (2001).

The last group of models consists of a combination of process- and organism-oriented models, which are seldom used because their applicability is limited by the data required to define the organism-oriented components (Smith et al., 1998). Some examples of combined models are proposed by O'Brien (1984) and Paustian et al. (1990).

In order to identify to what extent different models are suitable for long-term predictions of soil turnover, and to understand whether some simulators are more suited to specific environmental conditions or ecosystems, model comparisons were conducted using long-term experiments and multi-annual datasets. De Willigen (1991) tested 14 different models comparing their ability to simulate soil N turnover (e.g., mineralization and plant uptake). It was concluded that aboveground processes (e.g., plant growth) were easier to simulate than belowground transformations (e.g., soil water and N content), and that the more complex, multi-compartment models do not necessarily provide better results in terms of predictive capabilities. Rodrigo et al. (1997) compared the effects of soil moisture and temperature variations on nine different models (NCSOIL; SOILN; DAISY; Kersebaum's model, Kersebaum and Richter, 1991; MATHILD, Lafolie, 1991; TRITSIM, Mirschel et al., 1991; NLEAP, Shaffer et al., 1991; SUNDIAL, Bradbury et al., 1993; CANTIS, Neel, 1996) on predictions of soil C and N turnover. Not surprisingly, they observed the highest C decomposition and N mineralisation rates close to field capacity conditions and decreasing rates during soil drying. In this study good agreement between the different models for low moisture conditions was observed, whereas poor agreement was found in wet soils, with water saturation equal or above field capacity. A complete comparison of nine processoriented multi-compartment models (CANDY; NC-SOIL; RothC; DAISY; CENTURY, Parton et al., 1987; Verberne model, Verberne et al., 1990; ITE, Thornley, 1991; DNDC, Li et al., 1994; SOMM, Chertov and Komarov, 1997) was presented by Smith et al. (1997). A qualitative and quantitative evaluation of the performance of the models was carried out by comparing their ability to simulate observed data from seven different sites in temperate regions. A general conclusion of all these comparisons was that the errors derived from the tested models are not significantly different, meaning that the models provide consistent results except when a model is used for an application for which it was not developed. For example, the ITE and SOMM models were developed for grasslands while in the study of Smith et al. (1997) they were applied to crops. Model calibration is an additional source of uncertainties and makes the comparison of different models difficult. Pansu et al. (2004) presented

a qualitative comparison of the predictive performance of a family of five multi-compartment models, MOMOS-2 to -6, using <sup>14</sup>C- and <sup>15</sup>N-labelled species in field experiments. These models use the same conceptual approach but have different complexity, in that the number of compartments varies from 3 to 5 and the description of the biochemical transformation uses a different level of detail and simplification. Pansu et al. (2004) concluded that the simplifications do not decrease significantly model accuracy, but that the use of additional compartments results in improved long-term predictions.

Most of the currently available models are updated versions of earlier and original versions that have been modified to extend the applicability to specific ecosystems. Table 1 presents a list of the five most popular models, their main features together with the key references. The popular CENTURY model, originally devised for modelling soil nutrient dynamics in grassland systems, has been considerably modified since its first version. Smith et al. (1997) and Parton and Rassmussen (1994) modified the CENTURY model for application to crop and pasture systems, while Kelly et al. (1997), Peng et al. (1998) and Kirschbaum and Paul (2002) modified the model to be applied to forest ecosystems. Despite the ad hoc modifications, contrasting results in terms of predictive capabilities were obtained. The RothC model of Jenkinson et al. (1990), is an evolution of the model previously presented by Jenkinson and Rayner (1977), named Rothamsted. TOUGHREACT-N (Maggi et al., 2008) was developed to study the biogeochemical soil N cycle under different conditions of fertilization and irrigation. It is based on the multi-phase, multi-component reactive transport model TOUGHREACT (Xu et al., 2006), in turn an evolved and improved version of TOUGH2 (Pruess et al., 1999). SWIM (Krysanova et al., 1998), based on previously developed tools (MATSALU, Krysanova et al., 1989; SWAT, Arnold et al., 1993) and originally devised for modelling soil N cycle in mesoscale watersheds  $(10^2 \text{ to } 10^4 \text{ km}^2)$ , has recently been extended to better describe groundwater dynamics and processes in riparian zones (Hattermann et al., 2004; Wattenbach et al., 2005). FullCAM (Richards, 2001) accounts for full C turnover in forests, and is an integrated suite of sub-models: the empirical C tracking model CAMFor (Richards and Evans, 2000), the tree growth model 3PG (Landsberg and Waring, 1997), the

Model	Туре	Characteristics	References to updated versions
CENTURY Parton et al. (1987)	Process-oriented multi-compartment	<ul> <li>Long-term (decades to centuries) of SOM dynamics, plant growth and cycling of N, P and S, using monthly input data; Three organic compartments: active, slow and passive; Soil texture regulates the C transfer from one compartment to another.</li> </ul>	Parton and Rassmussen 1994 Fernandes et al. (1997) Kelly et al. (1997) Smith et al. (1997) Bolker et al. (1998) Peng et al. (1998) Kirschbaum and Paul (2002) Easter et al. (2007)
DAISY Hansen et al. (1991)	Process-oriented multi-compartment	<ul> <li>Simulates crop production, soil water and N dynamics;</li> <li>Four different modules: hydrological, soil temperature, soil nitrogen, and crop models;</li> <li>Decomposition rate constants depend on clay content;</li> <li>Vertical soil profile divided in layers considering physical and chemical characteristics.</li> </ul>	Jensen et al. (1997) Smith et al. (1997)
NCSOIL Molina et al. (1983)	Process-oriented multi-compartment	<ul> <li>Simulates C and N fluxes through soil microbes and organic components;</li> <li>Four organic compartments: plant residues, microbial biomass, humus and stable organic matter.</li> </ul>	Smith et al. (1997) Hadas et al. (2004)
CANDY (CArbon and Nitrogen DYnamics) Franko et al. (1995)	Process-oriented multi-compartment	<ul> <li>Composed of four different submodels: soil temperature, hydrological, crop and organic matter turnover models;</li> <li>Daily time step;</li> <li>Simulates soil N dynamics, temperature and moisture.</li> </ul>	Franko (1996) Franko et al. (1997) Smith et al. (1997)
RothC (Rothamsted model) Jenkinson and Rayner (1977)	Process-oriented multi-compartment	<ul> <li>N and C are not interconnected;</li> <li>C turnover sensitive to soil type, temperature, moisture and plant cover;</li> <li>Monthly input data.</li> </ul>	Smith et al. (1997) Easter et al. (2007) Kaonga and Coleman (2008) Jenkinson and Coleman (2008)

Table 1 Main characteristics of the five most popular models

litter decomposition model GENDEC (Moorhead and Reynolds, 1991) and the soil C turnover model RothC (Jenkinson, 1990). PASTIS (Lafolie, 1991; Garnier et al., 2001, 2003) is a one-dimensional mechanistic model that simulates the transport of water, solutes and heat using Richards' equation for water flow, the advection-dispersion equation for solute transport and the diffusion equation for heat flow. Some variations to this model have been implemented, such as PASTIS<sub>mulch</sub> (Findeling et al., 2007), which extends the original capabilities by including the physical effects of a surface residue mulch on rain interception and evaporation. Another example of model evolution is the family of models MOMOS-2 to -6, which are modified versions from the initial MOMOS-C (Sallih and Pansu, 1993) and MOMOS-N (Pansu et al., 1998) models. TRIPLEX (Peng et al., 2002) is a model of forest growth and C and N dynamics, and is a combination of three prior well-established models: 3PG (Landsberg and Waring, 1997), TREEDYN3.0 (Bossel, 1996) and CENTURY4.0 (Parton et al., 1993). Easter et al. (2007) developed a soil C modelling system, GEFSOC, aimed at modelling soil C stocks and exchange rates at regional or country scales in response to land use changes. The developed tool is based on three well-recognized models: the CENTURY general ecosystem model, the RothC soil C decomposition model and the empirical IPCC method (IPCC, 2003) for assessing soil C stock changes at regional scales. The model can be coupled to a soil and terrain digital database to include the topography and spatial soil variability of the studied area.

#### 5 Description of the C-N Model

The model we describe and use in this work was presented by Porporato et al. (2003). It belongs to the group of process-based models, with the soil organic matter and nutrients divided into five pools. Three pools consist of SOM (litter, humus and microbial biomass), while the remainder are for inorganic N. The model is applied to the root zone treated as a single unit, i.e., spatial variations are ignored.

The framework with three organic pools is in good agreement with Jenkinson (1990), who proposed that process-based models should have between two and four pools to obtain reliable results. These pools represent the main components of the system, and C and N concentrations correspond to average values over the rooting depth ( $Z_r$ ) (Rodriguez-Iturbe and Porporato, 2004). This simplification is justified because soils have often a uniform distribution of SOM and inorganic N over the whole rooting depth (Porporato et al., 2003). This is not true however for the uppermost soil layer, where organic residues tend to accumulate, and acts as a source of litter to the layers beneath.

Additionally, some other simplifications were made during the development. First, SOM decomposition rates are known to vary over orders of magnitude among the different components and, as already described, each functional group of organisms has specific and highly variable decomposition and mineralization rates. In the model however no distinction is made between different microbial populations. Rather, for each pool, a single, first-order kinetic rate is used, which represents an average transformation rate. This approach, although approximate, reduces the number of model parameters and therefore simplifies its calibration. Decomposition rates vary however among the different pools: litter has faster decomposition than the humus pool. The second approximation concerns the C/N ratio. As for the transformation rates, the model considers a single C/N ratio for each pool, again representing an average value. In this case, the litter C/N ratio can be computed, for example, as the weighted average of the C/N ratios of the different species, weighted by their relative amount in the ecosystem. Other than this, vegetation characteristics (maximum evapotranspiration, wilting point, incipient stress point, etc.) are assumed constant. This is an important simplification, since in previous sections it was pointed out that climatic conditions influence vegetation growth

and deposition of fresh organic matter. The advantage is that we reduce and simplify the external factors influencing C and N turnover to soil type and moisture content dynamics.

Model inputs are precipitation and litter fall rates, while on output the extent of soil respiration, plant uptake, transpiration and leaching are recovered. The amount and frequency of precipitation are the only climatic variables considered. Isothermal conditions are assumed, meaning that variations of the average daily temperature within the year are limited. This assumption is clearly not satisfied in many climatic regions (e.g., at high latitude). On the other hand, the model can still be applicable given that, during the unfavourable season (too high or low temperature), the moisture content becomes an additional limiting factor, thus inhibiting soil respiration and transformations.

The model of Porporato et al. (2003) is comprised of a set of coupled non-linear ordinary differential equations. Each equation describes the mass balance of C and N in the five pools. An overview of the reaction network is given in Fig. 3. Moreover, since the soil moisture is the key factor in this model, and influences the decomposition and turnover rates as outlined above, soil water variations are computed from the water balance at one point. In order to facilitate model understanding and comparison with previous works, here we use the same notation as in Porporato et al. (2003), D'Odorico et al. (2004) and Rodriguez-Iturbe and Porporato (2004).

The evolution of C in the litter, humus and biomass pools is given by:

$$\frac{dC_l}{dt} = ADD + BD - DEC_l, \tag{6}$$

$$\frac{dC_h}{dt} = r_h DEC_l - DEC_h,\tag{7}$$

$$\frac{dC_b}{dt} = (1 - r_h - r_r)DEC_l + (1 - r_r)DEC_h - BD,$$
(8)

where  $C_l$ ,  $C_h$  and  $C_b$  are the C concentrations in the litter, humus and biomass pools respectively [M L<sup>3</sup>], *ADD* is the external input of C to the system [M L<sup>-2</sup> T<sup>-1</sup>], *BD* is the recycling rate of decaying biomass in the litter pool [M L<sup>-3</sup> T<sup>-1</sup>], *DEC*<sub>l</sub> and *DEC*<sub>h</sub> are the C fluxes leaving the litter and humus pools due to microbial decomposition [M L<sup>-3</sup> T<sup>-1</sup>], while  $r_h$  and  $r_r$  are non-dimensional coefficients representing the fractions



Fig. 3 Schematic representation of soil C and N cycles (after Rodriguez-Iturbe and Porporato, 2004)

of decomposed organic C that go into the humus pool and to respiration, respectively.

The combination of equations (6)–(8) gives the overall C balance equation  $(C_{tot})$  in the system:

$$\frac{dC_{tot}}{dt} = ADD - r_r DEC_l - r_r DEC_h.$$
(9)

The flux of C between two pools is described by first-order kinetic equations ( $DEC_l$ ,  $DEC_h$  and BD), where the reaction rates ( $k_l$ ,  $k_h$  and  $k_d$ , respectively) are weighted averages of the decomposition rates of the different organic molecules. The first-order kinetic equations of C decomposition and microbial death for the litter, humus and biomass pool are:

$$DEC_l = \phi \ f_d(s) \ C_b \ k_l \ C_l, \tag{10}$$

$$DEC_h = \phi f_d(s) C_b k_h C_h, \qquad (11)$$

$$BD = C_b k_d, \tag{12}$$

where  $\phi$  is a non-dimensional factor that accounts for a possible reduction of the decomposition rate when the

litter is very poor in N (high C/N ratio) and the immobilization is not sufficient to integrate the required N by the bacteria. This factor has an important influence on the dynamics of the biomass evolution, and details on how it is defined and computed can be found in Porporato et al. (2003). The term  $f_d(s)$  is a non-dimensional parameter that describes soil moisture effects on decomposition:

$$f_d(s) = \begin{cases} \frac{s}{s_{fc}}, s \leq s_{fc}, \\ \frac{s_{fc}}{s}, s > s_{fc}, \end{cases}$$
(13)

where  $S_{fc}$  is the soil field capacity (water content held in soil after excess water drained away by gravity). The main model parameters are listed in Table 2.

The N balance in the litter, humus and biomass pools is computed from the C balance equations, scaled by the appropriate C/N ratio:

$$\frac{dN_l}{dt} = \frac{ADD}{(C/N)_{add}} + \frac{BD}{(C/N)_b} - \frac{DEC_l}{(C/N)_l},$$
 (14)

**Table 2**Summary of themain parameters consideredin the soil C-N model

Symbol	Representative model parameters	Dimensions
$C_l$	Carbon concentration in the litter pool	[M L <sup>-3</sup> ]
$C_h$	Carbon concentration in the humus pool	$[M L^{-3}]$
$C_b$	Carbon concentration in the biomass pool	$[M L^{-3}]$
N <sub>l</sub>	Organic nitrogen concentration in the litter pool	$[M L^{-3}]$
$N_h$	Organic nitrogen concentration in the humus pool	$[M L^{-3}]$
$N_b$	Organic nitrogen concentration in the biomass pool	$[M L^{-3}]$
$N^+$	Ammonium concentration in the soil	$[M L^{-3}]$
$N^{-}$	Nitrate concentration in the soil	$[M L^{-3}]$
ADD	External carbon input	$[M L^{-2}T^{-1}]$
BD	Carbon return to the litter pool due to death of microbial biomass	$[M L^{-1}]$
$DEC_l$	Carbon output due to microbial decomposition	$[M L^{-3}T^{-1}]$
$DEC_h$	Carbon output due to humus decomposition	$[M L^{-3}T^{-1}]$
r <sub>r</sub>	Portion of decomposing carbon that is lost by respiration	_
$r_h$	Fraction of decomposing litter which undergoes humification	_
$k_l$	Litter decomposition rate	$[L^{3}T^{-1}M^{-1}]$
$k_h$	Humus decomposition rate	$[L^{3}T^{-1}M^{-1}]$
$k_d$	Biomass death rate	$[T^{-1}]$
(C/N) <sub>add</sub>	Carbon to nitrogen ratio in the external added litter	_
(C/N) <sub>l</sub>	Carbon to nitrogen ratio in the litter pool	_
$(C/N)_h$	Carbon to nitrogen ratio in the humus pool	_
$(C/N)_b$	Carbon to nitrogen ratio in the biomass pool	_
MIN	Mineralization rate	$[T^{-1}M^{-1}L^3]$
NIT	Nitrification rate	$[T^{-1}M^{-1}L^3]$
$IMM^+$	Ammonium immobilization rate	$[M^{-1}L^3T^{-1}]$
$IMM^{-}$	Nitrate immobilization rate	$[M^{-1}L^{3}T^{-1}]$
$LE^+$	Ammonium leaching rate	$[M^{-1}L^3T^{-1}]$
$LE^{-}$	Nitrate leaching rate	$[M^{-1}L^{3}T^{-1}]$
$UP^+$	Ammonium plant uptake rate	$[M^{-1}L^{3}T^{-1}]$
$UP^-$	Nitrate plant uptake rate	$[M^{-1}L^{3}T^{-1}]$
Φ	Non-dimensional factor that takes into account the contribution due to either the net mineralization or to the immobilization	_
$\phi$	Non-dimensional factor that accounts for a possible reduction of the decomposition rate when the litter is very poor in N and the immobilization is not sufficient to integrate the required N by the bacteria	-
$f_d(s)$	Non-dimensional factor (soil moisture effect on decomposition)	-
S	Soil moisture	_
Zr	Rooting depth	[L]

$$\frac{dN_h}{dt} = r_h \frac{DEC_l}{(C/N)_h} - \frac{DEC_h}{(C/N)_h},$$
(15)

$$\frac{dN_b}{dt} = \left[1 - r_h \frac{(C/N)_l}{(C/N)_h}\right] \frac{DEC_l}{(C/N)_l} + \frac{DEC_h}{(C/N)_h} - \frac{BD}{(C/N)_b} - \Phi,$$
(16)

where  $N_l$ ,  $N_h$  and  $N_b$  are the N concentrations in the litter, humus and biomass pools, respectively [M L<sup>-3</sup> T<sup>-1</sup>], (C/N)<sub>add</sub>, (C/N)<sub>l</sub>, (C/N)<sub>h</sub> and (C/N)<sub>b</sub> are the C to N ratios of added organic matter, litter, humus and biomass pools, respectively, and  $\Phi$  is a term that takes into account the contribution due to either the net mineralization or to the immobilization [M L<sup>-3</sup> T<sup>-1</sup>]. This term relates the total mineralization and immobilization rates:

$$\Phi = MIN - IMM, \tag{17}$$

where *MIN* expresses the mineralization rate  $[M L^{-3}]$  $T^{-1}$  and *IMM* is the total rate of immobilization (sum of the N immobilization rate in the  $NH_4^+$  IMM<sup>+</sup> and  $NO_3^- IMM^-$  pools, respectively) [M L<sup>-3</sup> T<sup>-1</sup>]. When *IMM* is equal to zero, *MIN* is equal to  $\Phi$ , while when MIN is zero IMM is equal to  $-\Phi$ . The (C/N)<sub>b</sub> ratio is one of the most important parameters in the model, since the switch between mineralization and immobilization is defined in order to maintain as constant the C/N ratio of the biomass pool. If the organic matter is rich in N (and  $(C/N)_b$  is smaller than the value required to sustain growth of microbial biomass), decomposition results in surplus N. This is used by the microorganisms, and mineralization occurs. In contrast, if decomposition produces an environment poor in N, microorganisms will increase the immobilization rate of  $NH_4^+$  and  $NO_3^-$  in order to meet their requirements. This process is rather complex and very dynamic, as explained in Porporato et al. (2003).

N transfer between the pools is described by the same first-order kinetic transfer parameters used for C, with each term scaled by the corresponding C/N ratio (Fig. 3). The balance equations of inorganic N are:

$$\frac{dN^{+}}{dt} = MIN + IMM^{+} - NIT - LE^{+} - UP^{+}, \quad (18)$$
$$\frac{dN^{-}}{dt} = NIT - IMM^{-} - LE^{-} - UP^{-}, \quad (19)$$

where N<sup>+</sup> and N<sup>-</sup> are the inorganic N concentrations in the NH<sub>4</sub><sup>+</sup> and NO<sub>3</sub><sup>-</sup> pools, respectively [M L<sup>3</sup>], *NIT* is the nitrification rate [M L<sup>-3</sup> T<sup>-1</sup>],  $UP^+$  and  $UP^-$  are the N uptake by plants from the NH<sub>4</sub><sup>+</sup> and NO<sub>3</sub><sup>-</sup> pools, respectively [M L<sup>-3</sup> T<sup>-1</sup>], and  $LE^+$  and  $LE^-$  are N fluxes from the root zone towards the groundwater [M L<sup>-3</sup> T<sup>-1</sup>].

The combination of equations (14-16, 18, 19) gives the overall evolution of total N (N<sub>tot</sub>) in the system:

$$\frac{dN_{tot}}{dt} = \frac{ADD}{(C/N)_{add}} - LE^+ - UP^+ - LE^- - UP^-.$$
(20)

Equations (9) and (20) represent the total C and N sinks and sources of the system depicted in Fig. 3.

Although we have described the main elements of the model here, further descriptions – for example the rates of mineralization (*MIN*), immobilization (*IMM*<sup>+</sup> and *IMM*<sup>-</sup>), nitrification (*NIT*), plant uptake ( $UP^+$  and  $UP^-$ ) and leaching ( $LE^+$  and  $LE^-$ ) and their associated variables – can be found in Porporato et al. (2003), D'Odorico et al. (2004) and Rodriguez-Iturbe and Porporato (2004), together with additional discussion about the underlying assumptions and simplifications introduced in this model.

## 6 Modelling Scenarios

It has been shown in previous sections that land use change and hydroclimatic conditions are the main factors contributing to changes in soil C and N turnover. To test the relevance of mechanisms and parameters contributing to the fate of soil C and N, different modelling scenarios were simulated, for which the main variables are presented in Table 3. The combination of two different soils and two different climatic conditions gives four different scenarios, the results of which are presented subsequently. Due to the high frequency of  $NO_3^-$  variations and their importance to plant growth, D'Odorico et al. (2003) found that a daily temporal resolution was necessary to capture the impact of soil moisture on nutrient dynamics. A daily time step was used here also. The same initial C and N amounts in different pools, as well the same decomposition, mineralization and root uptake rates were considered in all scenarios, thereby allowing for a direct comparison between them. These values, presented in Table 4, were taken from D'Odorico et al. (2003).

## 6.1 Climate Scenarios

The occurrence and amount of precipitation are both intermittent and unpredictable. Precipitation scenarios were generated with a stochastic procedure described in Laio et al. (2001). Rainfall was assumed to follow a Poisson distribution with frequency  $\lambda$  $[T^{-1}]$ , and each rainfall event had infiltration sampled from an exponential distribution with mean  $\alpha$  [L]. Two different climates were considered with a different occurrence of precipitation. Rainfall interception by canopy depends on the vegetation type and structure and cannot be neglected, especially in arid areas

Table 3 Soil and clim	te parameters corresp	conding to the modelle	d scenarios
-----------------------	-----------------------	------------------------	-------------

			Climate A	Climate B
Climatic parameters			(seasonality)	(no seasonality)
Mean storm frequency $(\lambda)$		$d^{-1}$	0.3, 0.1	0.2
Mean storm [–] depth ( $\alpha$ )		m	$1.55 \times 10^{-2}, 4.8 \times 10^{-3}$	$1.08\times10^{-2}$
Soil parameters			Soil I (agricultural)	Soil II (forest)
Saturated hydraulic conduct	ivity (K)	${\rm m}~{\rm d}^{-1}$	1	2
Active root depth $(Z_r)$		m	0.3	0.7
Pore size distribution index	(b)	_	4.05	5.39
Porosity ( <i>n</i> )		-	0.35	0.45
Tortuosity ( <i>d</i> )		_	1.5	3
Field capacity $(s_{fc})$		-	0.35	0.55
Vegetation parameters	Maximum transpiration $(T_{max})$	m	$3.6 \times 10^{-3}$	$4.63 \times 10^{-3}$
	Canopy interception threshold	m	$5 \times 10^{-4}$	$2 \times 10^{-3}$
	Maximum evapotranspiration $(E_{max})$	$m d^{-1}$	$3.7 \times 10^{-3}$	$4.76 \times 10^{-3}$
	Evapotranspiration wilting point $(E_w)$	$m d^{-1}$	$10^{-4}$	$1.3 \times 10^{-4}$
	Hygroscopic point $(s_h)$	_	0.04	0.14
	Point of incipient stress (s*)	_	0.16	0.37
	Wilting point $(s_w)$	-	0.05	0.17
Modelled scenarios				
Scenario AI Agricultural soil with climatic seasonality				
Scenario BI	Agricultural soil without seasonality			
Scenario AII	Forest soil with climatic seasonality			
Scenario BII	Forest soil without seasonality			

where the evaporation rate can be significant. Canopy interception is accounted for in the model by defining a threshold value (e.g., high values for forests and low for grasslands) below which no rainfall reaches the soil surface. If instead the rainfall depth is higher than the threshold value, the total amount of rainfall reaching the soil surface is equal to the rainfall depth reduced by the canopy interception.

Parameters representing the two climates considered are presented in Table 3. Climate A is characterized by seasonality represented by two wet and two dry seasons over a year. This climate can be considered comparable to a Mediterranean climate, with two wet seasons, spring and fall (e.g., high  $\lambda$  and  $\alpha$ ) and two dry seasons, summer and winter (e.g., low  $\lambda$  and  $\alpha$ ). In contrast, climate B is characterized by a lack of seasonality, with relatively low but homogeneous amount of precipitation randomly distributed over the year, using  $\lambda$  and  $\alpha$  between those of the wet and dry seasons considered in climate A.

Although in Section 3 was pointed out that temperature exerts a control over soil C and N cycles, in this study only isothermal conditions were considered. This assumption was made for two reasons. First, the effect of temperature in many climates is less important than that of soil moisture and, second, because considering the effects of soil moisture alone the number of factors affecting soil nutrient cycles is reduced, and is therefore easier to understand the influence and feedbacks on soil changes and nutrient dynamics. Temperature variations are however closely related to climate conditions and therefore this factor should be considered in future analyses.

## 6.2 Soil Scenarios

We seek to identify whether different patterns of soil moisture evolve through time as a consequence of the combination of different processes in the soil-plantatmosphere system. To this end, equation (21) is used to calculate the soil moisture balance at a point (Laio et al., 2001):

$$nZ_r \frac{ds(t)}{dt} = R(t) - I(t) - Q[s(t);t] - E[s(t)] - L[s(t)],$$
(21)

**Table 4** Parameters related to carbon and nitrogen soil dynamics used in all model scenarios. Shaded parameters correspond to variables that are not focused upon in this manuscript. Their details can be found in Porporato et al. (2003)

Initial soil moisture	S	_	0.15
C litter pool	$C_l$	g m <sup>-3</sup>	1200
C humus pool	$C_h$	g m <sup>-3</sup>	8500
C biomass pool	$C_b$	$\mathrm{g}\mathrm{m}^{-3}$	50
N ammonium pool	$N^+$	g m <sup>-3</sup>	0
N nitrate pool	$N^{-}$	g m <sup>-3</sup>	1
Added litter	Add	$g C m^{-3} d^{-1}$	1.5
C/N ratio of added litter	$(C/N)_{add}$	_	58
C/N ratio of litter	$(C/N)_l$	_	22
C/N ratio of humus	$(C/N)_h$	_	22
C/N ratio of microbial biomass	$(C/N)_b$	-	11.5
C/N ratio of ammonium	$(C/N)^+$	-	1
C/N ration of nitrate	$(C/N)^{-}$	_	1
Isohumic coefficient	$r_h$	_	0.25
Respiration coefficient	r <sub>r</sub>	-	0.6
Litter decomposition rate	$k_l$	$m^3 d^{-1} g C^{-1}$	6.5 × 10 <sup>-</sup>
Factor of carbon return to litter pool	k <sub>d</sub>	d <sup>-1</sup>	$8.5 \times 10^{-3}$
Humus decomposition rate	$k_h$	$m^3 d^{-1} g C^{-1}$	$2.5 \times 10^{-6}$
Rate of nitrification	$k_n$	$m^3 d^{-1} g N^{-1}$	0.6
Ammonium immobilization coefficient	k <sub>amm</sub>	$m^3 d^{-1} g N^{-1}$	1
Nitrate immobilization coefficient	k <sub>nit</sub>	$m^3 d^{-1} g N^{-1}$	1
Ammonium plant demand	DEM <sup>+</sup>	$g N m^{-3} d^{-1}$	0.2
Nitrate plant demand	DEM <sup>-</sup>	$g N m^{-3} d^{-1}$	0.5
Fraction of dissolved ammonium	<i>a</i> +	-	0.05
Fraction of dissolved nitrate	<i>a</i> <sup>-</sup>	-	0.1

where *n* is the porosity;  $Z_r$  is the depth of active soil or root depth [L]; s(t) is the relative soil moisture content  $(0 \le s(t) \le 1)$ ; R(t) is the rainfall rate [L T<sup>-1</sup>]; I(t)is the amount of rainfall lost through interception by canopy cover [L T<sup>-1</sup>]; Q[s(t);t] is the rate of runoff [L T<sup>-1</sup>]; E[s(t)] is the evapotranspiration rate [L T<sup>-1</sup>]; and L[s(t)] is the leakage rate [L T<sup>-1</sup>]. The soil was assumed as a horizontal and homogeneous layer of depth  $Z_r$ . This is an important assumption because soil depth depends in time and space on two main parameters, soil structure and vegetation. In the simulations we considered the same soil depth for water balance and nutrient cycles. Water infiltration into the soil and runoff are entirely controlled by soil moisture dynamics, since water will infiltrate into the soil if there is available storage. Excess rainfall that cannot be stored in the soil is converted into runoff.

Although the vegetation type depends on both climate and soil, here the vegetation parameters were fixed for each soil in order to reduce the number of variables affecting changes in C and N fluxes (Table 3). This is also justified by the fact that vegetation parameters mostly depends on soil moisture, which directly depends on soil characteristics. Evapotranspiration varies from a maximum value  $E_{max}$  when soil moisture ranges between the maximum, unity, and the point of incipient stress, s\* (soil moisture level at which the plants begin to close stomata in response to water stress). The evapotranspiration rate decreases linearly from  $E_{max}$  to  $E_w$ , the latter rate corresponding to the soil moisture at the wilting point  $s_w$  (soil water content at which plants wilt and can no longer recover or, in terms of water potential, is defined as the suction head beyond which the plant can no longer take up water). Below this value, only transpiration is active, and the water loss rate is linear from  $E_w$  to zero at the point of hygroscopic water  $s_h$  (microscopic film of water covering soil particles not available for plants). More details and assumptions concerning evapotranspiration are given by Laio et al. (2001).

Verhoef and Brussaard (1990) defined a series of functional groups of pedofauna based on their contribution to nutrient decomposition and mineralization. Organisms belonging to the same functional group play a similar role in decomposition-mineralization transformations. For example, there is a functional group that includes organisms that pulverize, mix and granulate the soil. Such organisms are rather important because (i) they contribute to incorporate the organic residues available on the surface into the lower horizons; and (ii) they create large pores and channels that guarantee aeration of the soil profile and eliminate excess water. Other functional groups include pedofauna specialized in breaking down woody recalcitrant materials, in degrading litter and digesting organic residues, etc. (Brady and Weil, 2004). Although the functional

group concept is useful for modelling of soil nutrient cycles, here a more simplified approach considering a single value for nutrient decomposition and mineralization rates is used, representing the contribution of the entire pedofauna to these processes.

In combination with climates A and B, two soil types – named I and II – are considered, representative of agricultural and forest soils respectively (Table 3). Agricultural and forest soils have contrasting physical properties mainly due to management practices and the type of vegetation supported (Lutz and Chandler, 1955; Carmean, 1957; García-Oliva et al., 1994; de Moraes et al., 1996; Islam and Weil, 2000). The main differences between these soils are:

- Silt and clay content. Agricultural soils have lower amounts of silt and clay than natural forest soils, mostly as a result of preferential removal of these particles by water erosion (Islam and Weil, 2000).
- Soil aggregate stability. A higher input of litter fall combined with the absence of tillage and ploughing practices gives rise to forest soils with greater soil aggregate stability (Islam and Weil, 2000). Furthermore, forest soils are naturally protected from the impact of raindrops by the canopy and organic matter at the soil surface that absorbs raindrop energy (Carmean, 1957). In practice, the effect of raindrops is (i) removal of silt and clay particles and (ii) disruption of soil aggregates that subsequently can block large pores and reduce water percolation.
- *Bulk density and porosity*. Agricultural soils have higher bulk density and lower porosity than forest soils, mainly because of a greater residual sand content combined with poorer soil aggregation (García-Oliva et al., 1994; de Moraes et al., 1996).
- Soil structure. Agricultural soils often have a deteriorated structure in comparison to forest soils. This deterioration is apparent in pore modification, increased bulk density, increased compaction, and less stable aggregates (Carmean, 1957). In addition, compacted, impermeable layers or pans within the soil profile often develop as a consequence of repeated ploughing, mainly under wheel track patterns (Roger-Estrade et al., 2004; Coquet et al., 2005).
- Infiltration rate. As a consequence of above mentioned properties, which contribute to reduce the average pore size and their connectivity, the rate of infiltration is reduced in agricultural

soils. Additionally, forest vegetation has more extensive root networks, leading to large number of interconnected channels leading to rapid water infiltration (Lutz and Chandler, 1955).

 Runoff and soil erosion. Low infiltration rates of agricultural soils contribute to increased runoff, which emphasizes soil erosion and removal of siltclay soil particles (Lutz and Chandler, 1955).

Due to the above differences, the agricultural soil (soil I) is characterized by a relatively low saturated hydraulic conductivity (K), as well as lower values of pore size distribution (b), porosity (n) and soil field capacity ( $s_{fc}$ ) than soil II, representative of a forest soil (Ndiaye et al., 2007). Furthermore, the soil tortuosity is likely to be affected by the loss of structure and by the less extended root network of agricultural soils, the loss of connected porosity due to tillage processes and disturbed aggregates clogging large pores. The soil tortuosity index for the agricultural soil is also thus decreased in comparison with the forest soil (Table 3).

The rooting depth ( $Z_r$ ) considered is larger for forest soils than agricultural soils, since the root network is much more important for forest vegetation than agricultural. As previously mentioned, vegetation depends on both climate and soil. However, we have defined the vegetation parameters only in function of the soil type. As a consequence of the low C/N ratio of agricultural vegetation and microbial decomposers associated, cultivated soils typically have lower C/N ratio than forest soils (Zheng et al., 1999). Nevertheless, to facilitate the comparison between the four scenarios we assumed equal C/N ratios for both soils, as well as initial C and N concentrations.

#### 7 Results

Figures 4a and 4d present the precipitation over 20 y for climates A (seasonality) and B (no seasonality), respectively, while the evolution of water saturation for the same period – as computed with equation (21) – for each climate and soil type is depicted in Figs. 4b–4f, respectively. There is a marked difference in precipitation distribution between climates A and B, with wet and dry seasons in climate A and random uniformly distributed precipitation in climate B. Soil water saturation follows the dynamics imposed by



**Fig. 4** Simulated rainfall and corresponding water saturation of climate A (with seasonal effects) and climate B (without seasonal effects) for idealized agricultural (**a**, **b** and **c**) and forest (**d**, **e** and **f**) soils

precipitation, more notably in agricultural soils while the trend in water saturation evolution is smoothed in forest soils. As expected, peaks of soil saturation are lower for those soils under the influence of climate *B* than climate *A*, while water saturation in agricultural soils is lower than those of forest soils. The latter is due to different vegetation parameters associated with each soil type, specifically to the wilting point ( $s_w$ ), fixed at 0.05 and 0.17 for agricultural and forest soils, respectively. It is interesting to note that forest soils attenuate changes of soil saturation much more than agricultural soils (e.g., compare Figs. 4b to 4c or even Figs. 4e to 4f), and delays water saturation peaks, mostly due to difference rooting depth. The evolution of the five different nutrient pools, for agricultural and forest soils under the conditions of climate A, are presented in Fig. 5. Figures corresponding to the same pool are depicted with the same vertical scale in order to facilitate comparison between them. From the comparison is evident that seasonal effects are much more visible in agricultural than in forest soils. The lower rooting depth  $(Z_r)$ , soil hydraulic conductivity (K) and soil porosity (n) in the agricultural soil may be at the origin of these differences. Although forest soils present similar behaviour, small peaks observed in agricultural soils and corresponding to high frequency changes in precipitation, are not evident. As previously mentioned, litter decomposition



**Fig. 5** Simulated organic carbon and inorganic nitrogen concentrations for agricultural (left side) and forest (right side) soils under climate *A* conditions

in agricultural soils is enhanced by tillage and plough practices, represented in the model by lower values of saturated hydraulic conductivity (K), pore size distribution (b), porosity (n) and soil field capacity ( $s_{fc}$ ).

This explains the lower values of C litter pool in agricultural soils (compare Figs. 5a and 5f), although the high influence of seasonal climates over agricultural soils is likely to hide this fact. Furthermore, this trend is likely to influence the evolution of subsequent pools (Figs. 5g-5j). Peaks of litter C match well the biomass pool decrease, while decreases of litter C concentrations well correlate with the peaks of C concentration in the biomass pool for both agricultural and forest soils. This is not surprising since the augmentation of the biomass pool is linked to degradation of the litter pool. C concentrations in the humus pool  $(C_h)$ for agricultural and forest soils are lower than in other pools, since this pool is an intermediate pool between litter and biomass and there is almost no interaction with other variables. Concentrations of  $NH_4^+$  (Figs. 5d and 5i) show a trend similar to that of C in the litter pool, although  $NH_4^+$  variations are almost negligible since its concentration is very low. The NO<sub>3</sub> pool preserves much of the high-frequency variability imposed by the random forcing of precipitation, which is not surprising since  $NO_3^-$  dynamics are the final product of a number of intertwined processes in which both high- and low-frequency components interact. It is interesting to note the low levels of NO<sub>3</sub><sup>-</sup> obtained for agricultural soils (Fig. 5e) compared with the relatively high levels in forest soils (Fig. 5j). This is reflected in Fig. 6d and 6h, where  $NO_3^-$  leaching is almost nonexistent in forest soils while it is relatively important for agricultural soils. The higher rates of mineralization and NO<sub>3</sub> uptake in agricultural soils than in forest soils aid to explain this fact. Note that fertilization practices commonly undertaken in agricultural soils are not considered in the model. Thus, simulated results are in good agreement with the need of agricultural soils for regular fertilizer input, since high decomposition and mineralization rates (Figs. 6a and 6b),  $NO_3^-$  uptake by plants (Fig. 6c) and  $NO_3^-$  leaching (Fig. 6d) to lower layers results in low levels of  $NO_3^$ available for plants.

The results obtained, especially in forest soils, provide evidence of a structure with temporal amplitudes considerably larger than the one induced by the stochastic hydrologic forcing. This behaviour was already observed by Thornley et al. (1995) and D'Odorico et al. (2003). The presence of this behaviour in soil nutrient dynamics (see Figs. 5f–5j), a priori not related to climatic conditions, is a manifestation of the nonlinearity and degree of complexity of the whole system. Thornley et al. (1995) pointed out that such behaviour suggests that soil nutrients cycles could show cases of richer (perhaps chaotic) dynamics enmeshed with the variability that is directly induced by the stochastic hydrologic fluctuations. This issue is

not fully assessed yet to date, although a partial discussion is provided by Manzoni and Porporato (2007).

The evolution through time of the C and N pools in agricultural and forest soils under climate B (no seasonality) is presented in Fig. 7. A general trend for both agricultural and forest soil is that simulated concentrations in all C and N pools closely reproduce the concentrations observed in climate A (seasonal). This is supported by the similar decomposition rates between climate A (Fig. 6) and B (Fig. 8). Nevertheless, the absence of seasonality, which is reflected in agricultural and forest soils by a loss of the high frequency nutrient peaks, is finally translated in less nitrate leaching and less variability in the soil nutrients. While water saturation changes under seasonal climates contribute to hinder decomposition rates from time to time, this does not occur in climates without seasonal effects, where decomposition peaks are more continuous through time. The loss of climate seasonality affects neither the inverse trend between litter and biomass pools, nor the high frequency of  $NO_3^-$  changes in both agricultural and forest soils.

As observed under seasonal conditions, the frequency of changes in C and N concentrations is higher in agricultural than in forest soils, although the general trend remains quite similar for both soils. Again, C concentrations in the humus pool are relatively constant, and  $NH_4^+$  concentrations are low. Concentrations of  $NO_3^-$  are lower in agricultural soils than in forest soils, as observed under seasonal climate conditions. This fact supports what was mentioned in previous sections that agricultural soils are relatively poor in nutrients, independently of climate conditions. Furthermore, this is consistent with the higher mineralization rate (Fig. 8b), higher  $NO_3^-$  uptake by plants (Fig. 8c) and higher loss of  $NO_3^-$  by leaching (Fig. 8d) in agricultural than in forest soils.

Finally, some general insights can be gained from the model results. Great variations of the average value of C and N concentrations in forest soils are obtained both under seasonal and uniformly variable climatic conditions. These results show that a point measurement of litter, for example, is almost meaningless, and that time-series are needed to provide useful information for evaluation of soil nutrient turnover. It is worth noting also that steady state concentrations are not always reached, which means that longer simulations should be applied to better evaluate long term nutrient concentrations, mainly under non-seasonal conditions (see for example Fig. 5g in comparison to 5b).



**Fig. 6** Simulated rates of litter decomposition (a, e), net nitrogen mineralization (b, f), nitrate uptake (c, g) and nitrate leaching (d, h) for both agricultural and forest soils under climate A conditions

## 8 Conclusions

Land use changes affect soil properties and, thus, nutrient cycling dynamics. Changing soil properties means also changing the type of vegetation, altering even more the concomitant nutrient cycles. The example most widely observed is the change from forest to agricultural soils, due to increasing anthropogenic demands for food production. Agricultural practices generally cause changes in soil structure, compromising



**Fig. 7** Simulated organic carbon and inorganic nitrogen concentrations for agricultural (left side) and forest (right side) soils under climate *B* conditions

aggregation and porosity, leading to a soil structure decline. Tillage practices like mechanical mixing compact and reduce the size of aggregates and fills pore spaces with fines. The simulation of two different climates applied to two contrasting soil types (where specific vegetation properties were linked to each soil), showed that decomposition rates in agricultural soils are higher, which in addition to lower input of fresh



**Fig. 8** Simulated rates of litter decomposition (a, e), net nitrogen mineralization (b, f), nitrate uptake (c, g) and nitrate leaching (d, h) for both agricultural and forest soils under climate *B* conditions

litter over a year in comparison to a forested area, results in lower C and N concentrations. In consequence, less  $NO_3^-$  is available to plants, since plant  $NO_3^-$  uptake is higher and leaching is enhanced under these conditions. The leached  $NO_3^-$  is compensated for by the input of extra nutrients in agricultural soils via fertilization, a practice that, if poorly managed, contributes to diminish the quality of aquifers.

Model application under seasonal and non-seasonal climatic conditions resulted in a higher attenuation of punctual growing concentrations of soil C and N under the effect of a seasonal climate, as well as average concentrations higher than those under the effect of non-seasonal climates. Furthermore, the high frequency imposed by seasonal climates is attenuated in forest soils, while this high frequency is reflected in agricultural soil nutrient cycles. Nevertheless, the general trend of the temporal dynamic is similar under both seasonal and non-seasonal conditions.

Land use changes should include long term practices to avoid the loss of soil properties, contributing to the maintenance of optimal conditions for long term agricultural production. Crop rotation is an important management practice to avoid soil C losses following conversion from forest to agricultural land (Murty et al., 2002). Furthermore, Agboola (1981) and Bruce et al. (1999) proposed that a diminution of tillage processes minimizes soil erosion and decomposition rates, and thus soil C losses. The results presented here underscore that models dealing with soil nutrient turnover are potentially promising tools to design new soil practices and predict long-term effects of these practices on soil fertility.

Acknowledgements This research is part of the RECORD project of the Competence Centre Environment and Sustainability (CCES, http://www.cces.ethz.ch/projects/nature/Record). Funding has been provided by the Swiss National Science Foundation under grant 200021-113296.

## References

- Agboola A.A. (1981) The effects of different soil tillage and management practices on the physical and chemical properties of soil and maize yields in a rainforest zone of Western Nigeria, Agron. J. 73, 247–251.
- Agren G.I., McMurtrie R.E., Parton W.J., Pastor J., Shugart H.H. (1991) State-of-the-art of models of productiondecomposition linkages in conifer and grassland ecosystems, Ecol. Appl. 1, 118–138.
- Arnold J.G., Allen P.M., Bernhardt G. (1993) A comprehensive surface-groundwater flow model, J. Hydrol. 142, 47–69.
- Austin A.T., Yahdjian L., Stark J.M., Belnap J., Porporato A., Norton U., Ravetta D.A., Schaeffer S.M. (2004) Water pulses and biogeochemical cycles in arid and semiarid ecosystems, Oecologia 141, 221–235.
- Barbour M.G., Burk J.H., Pitts W.D., Gilliam F.S., Schwartz M.W. (1999) Terrestrial plant ecology. Pearson Benjamin Cummings, Menlo Park, Canada.
- Barry D.A., Prommer H., Miller C.T., Engesgaard P., Brun A., Zheng C. (2002) Modelling the fate of oxidisable organic contaminants in groundwater, Adv. Water Resour. 25, 945–983.
- Bell C., McIntyre N., Cox S., Tissue D., Zak J. (2008) Soil microbial responses to temporal variations of moisture and

temperature in a Chihuahuan desert grassland, Microb. Ecol. 56, 153–167.

- Berger T.W., Neubauer C., Glatzel G. (2002) Factors controlling soil carbon and nitrogen stores in pure stands of Norway spruce (Picea abies) and mixed species stands in Austria, For. Ecol. Manage. 159, 3–14.
- Bolker B.J., Pacala S.W., Parton W.J. (1998) Linear analysis of soil decomposition: insights from the century model, Ecol. Appl. 8, 425-439.
- Bonde T.A., Christensen B.T., Cerri C.C. (1992) Dynamics of soil organic matter as reflected by natural 13C abundance in particle size fractions of forested and cultivated oxisols, Soil Biol. Biochem. 24, 275–277.
- Borken W., Xu Y.-J., Davidson E.A., Beese F. (2002) Site and temporal variation of soil respiration in European beech, Norway spruce, and Scots pine forests, Glob. Change Biol. 8, 1205–1216.
- Borken W., Xu Y.J., Brumme R., Lamersdorf N. (1999) A Climate Change Scenario for Carbon Dioxide and Dissolved Organic Carbon Fluxes from a Temperate Forest Soil: Drought and Rewetting Effects, Soil Sci. Soc. Am. J. 63, 1848–1855.
- Bosatta E., Ågren G.I. (1991) Theoretical analysis of carbon and nutrient interactions in soils under energy-limited conditions, Soil Sci. Soc. Am. J. 55, 728–733.
- Bosatta E., Ågren G.I. (1994) Theoretical analysis of microbial biomass dynamics in soils, Soil Biol. Biochem. 26, 143–148.
- Bossel H. (1996) TREEDYN3 forest simulation model, Ecol. Model. 90, 187–227.
- Bradbury N.J., Whitmore A.P., Hart P.B.S., Jenkinson D.S. (1993) Modelling the fate of nitrogen in crop and soil in the years following application of <sup>15</sup>N-labelled fertilizer to winter wheat, J. Agric. Sci. 121, 363–379.
- Brady N.C., Weil R.R. (2004) Elements of the nature and properties of soils. Pearson Prentice Hall, Upper Saddle River, New Jersey, USA.
- Brams E. (1971) Continuous cultivation of West African soils: Organic matter diminution and effects of applied lime and phosphorus, Plant Soil 35, 401–414.
- Brown S., Lugo A. (1990) Effects of forest clearing and succession on the carbon and nitrogen content of soils in Puerto Rico and US Virgin Islands, Plant Soil 124, 53–64.
- Bruce J.P., Frome M., Haites E., Janzen H., Lal R., Paustian K. (1999) Carbon sequestration in soils, J. Soil Water Conserv. 54, 382–389.
- Brussaard L. (1998) Soil fauna, guilds, functional groups and ecosystem processes, Appl. Soil Ecol. 9, 123–135.
- Buchmann N. (2000) Biotic and abiotic factors controlling soil respiration rates in Picea abies stands, Soil Biol. Biochem. 32, 1625–1635.
- Carmean W.H. (1957) The structure of forest soils, Ohio J. Sci. 57, 165–168.
- Cherif M., Loreau M. (2009) When microbes and consumers determine the limiting nutrient of autotrophs: a theoretical analysis, Proc. R. Soc. B 276, 487–497.
- Chertov O.G., Komarov A.S. (1997) SOMM: a model of soil organic matter dynamics, Ecol. Model. 94, 177–189.
- Christensen B.T. (1996) Matching measurable soil organic matter fractions with conceptual pools in simulation models of carbon turnover: revision of model structure, in: Powlson D.S., Smith P., Smith J.U. (Eds.), Evaluation of soil organic matter models, Springer Verlag, Berlin, pp. 143–160.

- Christiansen J.R., Elberling B., Jansson P.E. (2006) Modelling water balance and nitrate leaching in temperate Norway spruce and beech forests located on the same soil type with the CoupModel, For. Ecol. Manage. 237, 545–556.
- Conant R.T., Klopatek J.M., Klopatek C.C. (2000) Environmental factors controlling soil respiration in three semiarid ecosystems, Soil Sci. Soc. Am. J. 64, 383–390.
- Coquet Y., Simunek J., Coutadeur C., van Genuchten M.T., Pot V., Roger-Estrade J. (2005) Water and solute transport in a cultivated silt loam soil. 2. Numerical analysis, Vadose Zone J. 4, 587–601.
- Csonka L.N. (1989) Physiological and genetic responses of bacteria to osmotic stress, Microbiol. Rev. 53, 121–147.
- Cui M., Caldwell M.M. (1997) A large ephemeral release of nitrogen upon wetting of dry soil and corresponding root responses in the field, Plant Soil 191, 291–299.
- Curiel Yuste J., Baldocchi D.D., Gershenson A., Goldstein A., Misson L., Wong S. (2007) Microbial soil respiration and its dependency on carbon inputs, soil temperature and moisture, Glob. Change Biol. 13, 2018–2035.
- D'Odorico P., Laio F., Porporato A., Rodriguez-Iturbe I. (2003) Hydrologic controls on soil carbon and nitrogen cycles. II. A case study, Adv. Water Resour. 26, 59–70.
- D'Odorico P., Porporato A., Laio F., Ridolfi L., Rodriguez-Iturbe I. (2004) Probabilistic modeling of nitrogen and carbon dynamics in water-limited ecosystems, Ecol. Model. 179, 205–219.
- Dalal R.C., Mayeer R.J. (1986) Long-term trends in fertility of soils under continuous cultivation and cereal cropping in Southern Queensland. I. Overall changes in soil properties and trends in winter cereal yield, Aust. J. Soil Res. 24, 265–279.
- Davidson E.A., Ackerman I.L. (1993) Changes in soil carbon inventories following cultivation of previously untilled soils, Biogeochemistry 20, 161–193.
- Davidson E.A., Belk E., Boone R.D. (1998) Soil water content and temperature as independent or confounded factors controlling soil respiration in a temperate mixed hardwood forest, Glob. Change Biol. 4, 217–227.
- Davidson E.A., Verchot L.V., Cattânio J.H., Ackerman I.L., Carvalho J.E.M. (2000) Effects of soil water content on soil respiration in forests and cattle pastures of eastern Amazonia, Biogeochemistry 48, 53–69.
- de Moraes J.F.L., Volkoff B., Cerri C.C., Bernoux M. (1996) Soil properties under Amazon forest and changes due to pasture installation in Rondônia, Brazil, Geoderma 70, 63–81.
- de Willigen P. (1991) Nitrogen turnover in the soil-crop system; comparison of fourteen simulation models, Nutr. Cycl. Agroecosyst. 27, 141–149.
- Desjardins T., Andreux F., Volkoff B., Cerri C.C. (1994) Organic carbon and 13C contents in soils and soil size-fractions, and their changes due to deforestation and pasture installation in eastern Amazonia, Geoderma 61, 103–118.
- Easter M., Paustian K., Killian K., Williams S., Feng T., Al-Adamat R., Batjes N.H., Bernoux M., Bhattacharyya T., Cerri C.C., Cerri C.E.P., Coleman K., Falloon P., Feller C., Gicheru P., Kamoni P., Milne E., Pal D.K., Powlson D.S., Rawajfih Z., Sessay M., Wokabi S. (2007) The GEFSOC soil carbon modelling system: A tool for conducting regionalscale soil carbon inventories and assessing the impacts of

land use change on soil carbon, Agric. Ecosyst. Environ. 122, 13–25.

- Eckersten H., Beier C. (1998) Comparison of N and C dynamics in two Norway spruce stands using a process oriented simulation model, Environ. Pollut. 102, 395–401.
- Eckersten H., Blombäck K., Kätterer T., Nyman P. (2001) Modelling C, N, water and heat dynamics in winter wheat under climate change in southern Sweden, Agric. Ecosyst. Environ. 86, 221–235.
- Ellert B.H., Gregorich E.G. (1996) Storage of carbon, nitrogen and phosphorus in cultivated and adjacent forested soils of Ontario, Soil Sci. 161, 587–603.
- Elliott E.T., Paustian K., Frey S.D. (1996) Modeling the measurable or measuring the modelable: a hierarchical approach to isolating meaningful soil organic matter fractions, in: Powlson D.S., Smith P., Smith J.U. (Eds.), Evaluation of soil organic matter models. Using existing long-term datasets, Springer, Heidelberg, p. 429.
- Epron D., Nouvellon Y., Roupsard O., Mouvondy W., Mabiala A., Saint-André L., Joffre R., Jourdan C., Bonnefond J.-M., Berbigier P., Hamel O. (2004) Spatial and temporal variations of soil respiration in a Eucalyptus plantation in Congo, For. Ecol. Manage. 202, 149–160.
- Feigl B., Melillo J., Cerri C. (1995) Changes in the origin and quality of soil organic matter after pasture introduction in Rondônia (Brazil), Plant Soil 175, 21–29.
- Feller C., Beare M.H. (1997) Physical control of soil organic matter dynamics in the tropics, Geoderma 79, 69–116.
- Fenchel T., King G.M., Blackburn T.H. (1998) Bacterial biogeochemistry. The ecophysiology of mineral cycling, Academic Press, San Diego, California, USA.
- Fernandes E.C.M., Motavalli P.P., Castilla C., Mukurumbira L. (1997) Management control of soil organic matter dynamics in tropical land-use systems, Geoderma 79, 49–67.
- Findeling A., Garnier P., Coppens F., Lafolie F., Recous S. (2007) Modelling water, carbon and nitrogen dynamics in soil covered with decomposing mulch, Eur. J. Soil Sci. 58, 196–206.
- Franko U. (1996) Modelling approaches of soil organic matter within the CANDY system, in: Powlson D.S., Smith P., Smith J.U. (Eds.), Evaluation of soil organic matter models using existing, long-term datasets, NATO ASI, I38, Springer, Berlin, pp. 247–254.
- Franko U., Crocker G.J., Grace P.R., Klír J., Körschens M., Poulton P.R., Richter D.D. (1997) Simulating trends in soil organic carbon in long-term experiments using the CANDY model, Geoderma 81, 109–120.
- Franko U., Oelschlägel B., Schenk S. (1995) Simulation of temperature, water- and nitrogen dynamics using the model CANDY, Ecol. Model. 81, 213–222.
- Frolking S., Roulet N.T., Moore T.R., Richard P.J.H., Lavoie M., Muller S.D. (2001) Modeling northern peatland decomposition and peat accumulation, Ecosyst. 4, 479–498.
- Furniss P.R., Ferrar P., Morris J.W., Bezuidenhout J.J. (1982) A model of savannah litter decomposition, Ecol. Model. 17, 33–51.
- García-Oliva F., Casar I., Morales P., Maass J.M. (1994) Forest-to-pasture conversion influences on soil organic carbon dynamics in a tropical deciduous forest, Oecologia 99, 392–396.

- García-Oliva F., Sanford R.L., Kelly E. (1999) Effects of slash-and-burn management on soil aggregate organic C and N in a tropical deciduous forest, Geoderma 88, 1–12.
- Garnier P., Néel C., Aita C., Recous S., Lafolie F., Mary B. (2003) Modelling carbon and nitrogen dynamics in a bare soil with and without straw incorporation, Eur. J. Soil Sci. 54, 555–568.
- Garnier P., Néel C., Mary B., Lafolie F. (2001) Evaluation of a nitrogen transport and transformation model in a bare soil, Eur. J. Soil Sci. 52, 253–268.
- Gignoux J., House J., Hall D., Masse D., Nacro H.B., Abbadie L. (2001) Design and test of a generic cohort model of soil organic matter decomposition: the SOMKO model, Glob. Ecol. Biogeogr. 10, 639–660.
- Goulden M.L., Miller S.D., da Rocha H.R., Menton M.C., de Freitas H.C., e Silva Figueira A.M., de Sousa C.A.D. (2004) Diel and seasonal patterns of tropical forest CO<sub>2</sub> exchange, Ecol. Appl. 14, 42–54.
- Groffman P.M., Boulware N.J., Zipperer W.C., Pouyat R.V., Band L.E., Colosimo M.F. (2002) Soil nitrogen cycle processes in urban riparian zones, Environ. Sci. Technol. 36, 4547–4552.
- Hadas A., Kautsky L., Goek M., Erman Kara E. (2004) Rates of decomposition of plant residues and available nitrogen in soil, related to residue composition through simulation of carbon and nitrogen turnover, Soil Biol. Biochem. 36, 255–266.
- Hansen S., Jensen H.E., Nielsen N.E., Svendsen H. (1991) Simulation of nitrogen dynamics and biomass production in winter wheat using the Danish simulation model DAISY, Fert. Res. 27, 245–259.
- Hattermann F., Krysanova V., Wechsung F., Wattenbach M. (2004) Integrating groundwater dynamics in regional hydrological modelling, Environ. Model. Softw. 19, 1039–1051.
- Henriksen T.M., Breland T.A. (1999) Evaluation of criteria for describing crop residue degradability in a model of carbon and nitrogen turnover in soil, Soil Biol. Biochem. 31, 1135–1149.
- Houghton R.A. (2002) The global effects of tropical deforestation, Environ. Sci. Technol. 24, 414–422.
- Houghton R.A., Lefkowitz D.S., Skole D.L. (1991) Changes in the landscape of Latin America between 1850 and 1985 I. Progressive loss of forests, For. Ecol. Manage. 38, 143–172.
- Howard D.M., Howard P.J.A. (1993) Relationships between CO<sub>2</sub> evolution, moisture content and temperature for a range of soil types, Soil Biol. Biochem. 25, 1537–1546.
- Hunt H.W., Trlica M.J., Redente E.F., Moore J.C., Detling J.K., Kittel T.G.F., Walter D.E., Fowler M.C., Klein D.A., Elliott E.T. (1991) Simulation model for the effects of climate change on temperate grassland ecosystems, Ecol. Model. 53, 205–246.
- IPCC (2003) Good practice guidance for land use, land use change and forestry, in: Penman J., Gytarsky M., Hiraishi T., Krug T., Kruger T., Pipatti R., Buendia L., Miwa K., Ngara T., Tanabe K., Wagner F. (Eds.), Intergovernmental Pannel on Climate Change, Hayama, Kanagawa, Japan, p. 599.
- Islam K.R., Kamaluddin M., Bhuiyan M.K., Badruddin A. (1999) Comparative performance of exotic and indigenous forest species for tropical semievergreen degraded forest land

reforestation in Chittagong, Bangladesh, Land Degrad. Dev. 10, 241–249.

- Islam K.R., Weil R.R. (2000) Land use effects on soil quality in a tropical forest ecosystem of Bangladesh, Agric. Ecosyst. Environ. 79, 9–16.
- Ito A. (2007) Simulated impacts of climate and land-cover change on soil erosion and implication for the carbon cycle, 1901 to 2100, Geophys. Res. Lett. 34.
- Jenkinson D.S. (1990) The turnover of organic carbon and nitrogen in soil, Philos. Trans. R. Soc. Lond. B 329, 361–368.
- Jenkinson D.S., Adams D.E., Wild A. (1991) Model estimates of CO<sub>2</sub> emissions from soil in response to global warming, Nature 351, 304–306.
- Jenkinson D.S., Coleman K. (2008) The turnover of organic carbon in subsoils. Part 2. Modelling carbon turnover, Eur. J. Soil Sci. 59, 400–413.
- Jenkinson D.S., Rayner J.H. (1977) The turnover of soil organic matter in some of the Rothamsted classical experiments, Soil Sci. 123, 298–305.
- Jensen L.S., Mueller T., Nielsen N.E., Hansen S., Crocker G.J., Grace P.R., Klír J., Körschens M., Poulton P.R. (1997) Simulating trends in soil organic carbon in long-term experiments using the soil-plant-atmosphere model DAISY, Geoderma 81, 5–28.
- Johnson D.W. (1992) Effects of forest management on soil carbon storage, Water Air Soil Pollut. 64, 83–120.
- Johnsson H., Bergstrom L., Jansson P.-E., Paustian K. (1987) Simulated nitrogen dynamics and losses in a layered agricultural soil, Agric. Ecosyst. Environ. 18, 333–356.
- Juo A.S., Lal R. (1979) Nutrient profile in a tropical alfisol under conventional and no-till systems, Soil Sci. 127, 168–173.
- Kaonga M.L., Coleman K. (2008) Modelling soil organic carbon turnover in improved fallows in eastern Zambia using the RothC-26.3 model, For. Ecol. Manage. 256, 1160–1166.
- Kelly R.H., Parton W.J., Crocker G.J., Graced P.R., Klír J., Körschens M., Poulton P.R., Richter D.D. (1997) Simulating trends in soil organic carbon in long-term experiments using the century model, Geoderma 81, 75–90.
- Kersebaum K.C., Richter J. (1991) Modelling nitrogen dynamics in a plant-soil system with a simple model for advisory purposes, Nutr. Cycl. Agroecosyst. 27, 273–281.
- Kieft T.L., soroker E., Firestone M.K. (1987) Microbial biomass response to a rapid increase in water potential when dry soil is wetted, Soil Biol. Biochem. 19, 119–126.
- Kieft T.L., White C.S., Loftin S.R., Aguilar R., Craig J.A., Skaar D.A. (1998) Temporal dynamics in soil carbon and nitrogen resources at a grassland-shrubland ecotone, Ecology 79, 671–683.
- Kiese R., Butterbach-Bahl K. (2002) N<sub>2</sub>O and CO<sub>2</sub> emissions from three different tropical forest sites in the wet tropics of Queensland, Australia, Soil Biol. Biochem. 34, 975–987.
- Killham K., Foster R. (1994) Soil ecology. Cambridge University Press, Cambridge, UK.
- Kirschbaum M.U.F., Paul K.I. (2002) Modelling C and N dynamics in forest soils with a modified version of the CEN-TURY model, Soil Biol. Biochem. 34, 341–354.
- Koch O., Tscherko D., Kandeler E. (2007) Temperature sensitivity of microbial respiration, nitrogen mineralization, and potential soil enzyme activities in organic alpine soils, Glob. Biogeochem. Cycles 21.

- Krysanova V., Meiner A., Roosaare J., Vasilyev A. (1989) Simulation modelling of the coastal waters pollution from agricultural watershed, Ecol. Model. 49, 7–29.
- Krysanova V., Müller-Wohlfeil D.-I., Becker A. (1998) Development and test of a spatially distributed hydrological/water quality model for mesoscale watersheds, Ecol. Model. 106, 261–289.
- Kuijper L.D.J., Berg M.P., Morriën E., Kooi B.W., Verhoef H.A. (2005) Global change effects on a mechanistic decomposer food web model, Glob. Change Biol. 11, 249–265.
- Lafolie F. (1991) Modelling water flow, nitrogen transport and root uptake including physical non-equilibrium and optimization of the root water potential, Fert. Res. 27, 215–231.
- Laio F., Porporato A., Ridolfi L., Rodriguez-Iturbe I. (2001) Plants in water-controlled ecosystems: active role in hydrological processes and response to water stress. II. Probabilistic soil moisture dynamics, Adv. Water Resour. 24, 707–723.
- Landsberg J.J., Waring R.H. (1997) A generalised model of forest productivity using simplified concepts of radiation-use efficiency, carbon balance and partitioning, For. Ecol. Manage. 95, 209–228.
- Li C., Frolking S., Harriss R. (1994) Modeling carbon biogeochemistry in agricultural soils, Glob. Biogeochem. Cycles 8, 237–254.
- Lugo A., Brown S. (1993) Management of tropical soils as sinks or sources of atmospheric carbon, Plant Soil 149, 27–41.
- Lutz H.J., Chandler R.F. (1955) Forest soils. John Willey, New York, US.
- Maggi F., Gu C., Riley W.J., Hornberger G.M., Venterea R.T., Xu T., Spycher N., Steefel C., Miller N.L., Oldenburg C.M. (2008) A mechanistic treatment of the dominant soil nitrogen cycling processes: model development, testing, and application, J. Geophys. Res. 113.
- Mann L.K. (1986) Changes in soil carbon storage after cultivation, Soil Sci. 142, 279–288.
- Manzoni S., Porporato A. (2007) A theoretical analysis of nonlinearities and feedbacks in soil carbon and nitrogen cycles, Soil Biol. Biochem. 39, 1542–1556.
- Manzoni S., Porporato A. (2009) Soil carbon and nitrogen mineralization: Theory and models across scales, Soil Biol. Biochem. 41, 1355–1379.
- Martins P.F.S., Cerri C.C., Volkoff B., Andreux F., Chauvel A. (1991) Consequences of clearing and tillage on the soil of a natural Amazonian ecosystem, For. Ecol. Manage. 38, 273–282.
- Matus F.J., Rodríguez J. (1994) A simple model for estimating the contribution of nitrogen mineralization to the nitrogen supply of crops from a stabilized pool of soil organic matter and recent organic input, Plant Soil 162, 259–271.
- Melillo J.M. (1996) Carbon and nitrogen interactions in the terrestrial biosphere: Anthropogenic effects, in: Walker B., W. Steffen (Eds.), Gl. Ch. Ter. Ecosyst., Cambridge University Press, Cambridge, UK, pp. 431–450.
- Mirschel W., Kretschmer H., Matthäus E., Koitzsch R. (1991) Simulation of the effects of nitrogen supply on yield formation processes in winter wheat with the model TRITSIM, Fert. Res. 27, 293–304.
- Misson L., Tang J., Xu M., McKay M., Goldstein A. (2005) Influences of recovery from clear-cut, climate variability, and thinning on the carbon balance of a young ponderosa pine plantation, Agric. For. Meteorol. 130, 207–222.

- Molina J.A.E., Clapp C.E., Shaffer M.J., Chichester F.W., Larson W.E. (1983) NCSOIL, a model of nitrogen and carbon transformations in soil: description, calibration and behavior, Soil Sci. Soc. Am. J. 47, 85–91.
- Moore J.C., Berlow E.L., Coleman D.C., Ruiter P.C., Dong Q., Hastings A., Johnson N.C., McCann K.S., Melville K., Morin P.J., Nadelhoffer K., Rosemond A.D., Post D.M., Sabo J.L., Scow K.M., Vanni M.J., Wall D.H. (2004) Detritus, trophic dynamics and biodiversity, Ecol. Lett. 7, 584–600.
- Moore J.C., McCann K., de Ruiter P.C. (2005) Modeling trophic pathways, nutrient cycling, and dynamic stability in soils, Pedobiologia 49, 499–510.
- Moore T.R., Trofymow J.A., Siltanen M., Kozak L.M. (2008) Litter decomposition and nitrogen and phosphorous dynamics in peatlands and uplands over 12 years in central Canada, Oecologia 157, 317–325.
- Moorhead D.L., Currie W.S., Rastetter E.B., Parton W.J., Harmon M.E. (1999) Climate and litter quality controls on decomposition: an analysis of modeling approaches, Glob. Biogeochem. Cycles 13, 575–589.
- Moorhead D.L., Reynolds J.F. (1991) A general model of litter decomposition in the northern Chihuahuan Desert, Ecol. Model. 56, 197–219.
- Motavalli P.P., Discekici H., Kuhn J. (2000) The impact of land clearing and agricultural practices on soil organic C fractions and CO<sub>2</sub> efflux in the Northern Guam aquifer, Agric. Ecosyst. Environ. 79, 17–27.
- Murty D., M.Kirschbaum U.F., McMurtrie R.E., McGilvray H. (2002) Does conversion of forest to agricultural land change soil carbon and nitrogen? A review of the literature, Glob. Change Biol. 8, 105–123.
- Ndiaye B., Molénat J., Hallaire V., Gascuel C., Hamon Y. (2007) Effects of agricultural practices on hydraulic properties and water movement in soils in Brittany (France), Soil Till. Res. 93, 251–263.
- Neel C. (1996) Modélisation couplée du transfert et des transformations de l'azote: Paramétrisation et évaluation d'un modèle en sol nu, Université Pierre et Marie Curie, Paris, p. 276.
- Nicolardot B., Recous S., Mary B. (2001) Simulation of C and N mineralisation during crop residue decomposition: a simple dynamic model based on the C:N ratio of the residues, Plant Soil 228, 83–103.
- Nye P.H., Greenland D.J. (1964) Changes in the soil after clearing tropical forest, Plant Soil 21, 101–112.
- O'Brien B.J. (1984) Soil organic carbon fluxes and turnover rates estimated from radiocarbon enrichments, Soil Biol. Biochem. 16, 115–120.
- Pansu M., Bottner P., Sarmiento L., Metselaar K. (2004) Comparison of five soil organic matter decomposition models using data from a <sup>14</sup>C and <sup>15</sup>N labeling field experiment, Glob. Biogeochem. Cycles 18, GB4022.
- Pansu M., Sallih Z., Bottner P. (1998) Modelling of soil nitrogen forms after organic amendments under controlled conditions, Soil Biol. Biochem. 30, 19–29.
- Pansu M., Sarmiento L., Metselaar K., Hervé D., Bottner P. (2007) Modelling the transformations and sequestration of soil organic matter in two contrasting ecosystems of the Andes, Eur. J. Soil Sci. 58, 775–785.

- Parton W.J., Rassmussen P.E. (1994) Long-term effects of crop management in wheat-fallow: II. CENTURY model simulations, Soil Sci. Soc. Am. J. 58, 530–536.
- Parton W.J., Schimmel D.S., Cole C.V., Ojima D.S. (1987) Analysis of factors controlling soil organic matter levels in great plains grasslands, Soil Sci. Soc. Am. J. 51, 1173–1179.
- Parton W.J., Scurlock J.M.O., Ojima D.S., Gilmanov T.G., Scholes R.J., Schimel D.S., Kirchner T., Menaut J.C., Seastedt T., E. Garcia Moya, A. Kamnalrut, Kinyamario J.I. (1993) Observations and modeling of biomass and soil organic matter dynamics for the frassland biome worldwide, Glob. Biogeochem. Cycles 7, 785–809.
- Pastor J., Post W. (1986) Influence of climate, soil moisture, and succession on forest carbon and nitrogen cycles, Biogeochemistry 2, 3–27.
- Paul E.A. (1976) Carbon, nitrogen, phosphorous, sulfur and selenium cycles, Proc. 2nd Int. Symp. Environ. Biogechem.
- Paul K.I., Polglase P.J. (2004) Prediction of decomposition of litter under eucalypts and pines using the FullCAM model, For. Ecol. Manage. 191, 73–92.
- Paustian K., Andren O., Clarholm M., Hansson A.C., Johansson G., J. Lagerlof, Lindberg T., Pettersson R., Sohlenius B. (1990) Carbon and nitrogen budgets of four agroecosystems with annual and perennial crops, with and without N fertilization, J. Appl. Ecol. 27, 60–84.
- Peng C., Apps M.J., Price D.T., Nalder I.A., Halliwell D.H. (1998) Simulating Carbon Dynamics Along the Boreal Forest Transect Case Study (BFTCS) in Central Canada 1. Model Testing, Glob. Biogeochem. Cycles 12, 381–402.
- Peng C., Liu J., Dang Q., Apps M.J., Jiang H. (2002) TRIPLEX: a generic hybrid model for predicting forest growth and carbon and nitrogen dynamics, Ecol. Model. 153, 109–130.
- Persson T. (1983) Influence of soil animals on nitrogen mineralisation in a northern Scots pine forest, in: Lebrun P., Andre H.M., A. de Medts, C. Gregoire-Wibo, G. Wauthy (Eds.), Proc. of the VIII Int. Colloquium of soil Zoology, Louvain-la-Neuve, Belgium.
- Pietikåinen J., Pettersson M., Bååth E. (2005) Comparison of temperature effects on soil respiration and bacterial and fungal growth rates, FEMS Microbiol. Ecol. 52, 49–58.
- Porporato A., D'Odorico P., Laio F., Rodriguez-Iturbe I. (2003) Hydrologic controls on soil carbon and nitrogen cycles. I. Modelling scheme, Adv. Water Resour. 26, 45–58.
- Post J., Krysanova V., Suckow F., Mirschel W., Rogasik J., Merbach I. (2007) Integrated eco-hydrological modelling of soil organic matter dynamics for the assessment of environmental change impacts in meso- to macro-scale river basins, Ecol. Model. 206, 93–109.
- Post W.M., Emanuel W.R., Zinke P.J., Stangenberger A.G. (1982) Soil carbon pools and world life zones, Nature 298, 156–159.
- Prasad P., Basu S., Behera N. (1995) A comparative account of the microbiological characteristics of soils under natural forest, grassland and cropfield from Eastern India, Plant Soil 175, 85–91.
- Prober S.M., Thiele K.R., Koen L.T.B. (2005) Restoring ecological function in temperate grassy woodlands: manipulating soil nutrients, exotic annuals and native perennial grasses through carbon supplements and spring burns, J. Appl. Ecol. 42, 1073–1085.

- Pruess K., Oldenburg C.M., Moridis G.J. (1999) TOUGH2 user's guide version 2, Lawrence Berkeley National Laboratory, University of California, Berkeley, California, US, p. 192.
- Reichstein M., Rey A., Freibauer A., Tenhunen J., Valentini R., Banza J., Casals P., Cheng Y., GrÃ<sup>1</sup>/<sub>4</sub>nzweig J.M., Irvine J., Joffre R., Law B.E., Loustau D., Miglietta F., Oechel W., Ourcival J.-M., Pereira J.S., Peressotti A., Ponti F., Qi Y., Rambal S., Rayment M., Romanya J., Rossi F., Tedeschi V., Tirone G., Xu M., Yakir D. (2003) Modeling temporal and large-scale spatial variability of soil respiration from soil water availability, temperature and vegetation productivity indices, Glob. Biogeochem. Cycles 17, 1104.
- Reicosky D.C., Dugas W.A., Torbert H.A. (1997) Tillageinduced soil carbon dioxide loss from different cropping systems, Soil Till. Res. 41, 105–118.
- Reiners W.A., Bouwman A.F., Parsons W.F.J., Keller M. (1994) Tropical rain forest conversion to pasture: changes in vegetation and soil properties, Ecol. Appl. 4, 363–377.
- Rey A., Pegoraro E., Tedeschi V., I. De Parri, Jarvis P., Valentini R. (2002) Annual variation in soil respiration and its components in a coppice oak forest in Central Italy, Glob. Change Biol. 8, 851–866.
- Richards G.P. (2001) The FullCAM carbon accounting model: development, calibration and implementation for the national carbon accounting system, Australian Greenhouse Office, Canberra, Australia, pp. 6–27.
- Richards G.P., Evans D. (2000) Carbon accounting model for forests (CAMFor). User manual version 3.5, Australian Greenhouse Office, Canberra, Australia, p. 64.
- Rodrigo A., Recous S., Neel C., Mary B. (1997) Modelling temperature and moisture effects on C-N transformations in soils: comparison of nine models, Ecol. Model. 102, 325– 339.
- Rodriguez-Iturbe I., Porporato A. (2004) Ecohydrology of water-controlled ecosystems. Soil moisture and plant dynamics, Cambridge University Press, Cambridge, UK.
- Rodriguez-Iturbe I., Porporato A., Laio F., Ridolfi L. (2001) Plants in water-controlled ecosystems: active role in hydrologic processes and response to water stress: I. Scope and general outline, Adv. Water Resour. 24, 695–705.
- Roger-Estrade J., Richard G., Caneill J., Boizard H., Coquet Y., Defossez P., Manichon H. (2004) Morphological characterisation of soil structure in tilled fields: from a diagnosis method to the modelling of structural changes over time, Soil Till. Res. 79, 33–49.
- Rustad L.E., Huntington T.G., Boone R.D. (2000) Controls on soil respiration: Implications for climate change, Biogeochemistry 48, 1–6.
- Ryel R.J., Leffler A.J., Peek M.S., Ivans C.Y., Caldwell M.M. (2004) Water conservation in Artemisia tridentata through redistribution of precipitation, Oecologia 141, 335–345.
- Sallih Z., Pansu M. (1993) Modelling of soil carbon forms after organic amendment under controlled conditions, Soil Biol. Biochem. 25, 1755–1762.
- Savage K.E., Davidson E.A. (2001) Interannual variation of soil respiration in two New England forests, Glob. Biogeochem. Cycles 15, 337–350.
- Schimel D.S., Braswell B.H., Holland E.A., McKeown R., Ojima D.S., T.H. Painter, Parton W.J., Townsend A.R. (1994) Climatic, Edaphic, and Biotic Controls Over Storage and

Turnover of Carbon in Soils, Glob. Biogeochem. Cycles 8, 279–293.

- Schinner F., Kandeler E., Öhlinger R., Margesin R. (1995) Methods in soil biology, Springer-Verlag, Berlin, New York.
- Schlesinger W.H., Andrews J.A. (2000) Soil Respiration and the Global Carbon Cycle, Biogeochemistry 48, 7–20.
- Scholes M.C., Powlson D., Tian G. (1997) Input control of organic matter dynamics, Geoderma 79, 25–47.
- Schwinning S., Sala O.E. (2004) Hierarchy of responses to resource pulses in arid and semi-arid ecosystems, Oecologia 141, 211–220.
- Schwinning S., Starr B.I., Ehleringer J.R. (2003) Dominant cold desert plants do not partition warm season precipitation by event size, Oecologia 136, 252–260.
- Scott-Denton L.E., Rosenstiel T.N., Monson R.K. (2006) Differential controls by climate and substrate over the heterotrophic and rhizospheric components of soil respiration, Glob. Change Biol. 12, 205–206.
- Shaffer M.J., Halvorson A.D., Pierce F.J. (1991) Nitrate leaching and economic analysis package (NLEAP): model description and application, in: Follet R.F. (Ed.), Managing nitrogen for groundwater quality and farm profitability, ASA, Madison, Wisconsin, US, pp. 285–232.
- Smil V. (1999) Crop residues: agriculture's largest Harvest, Bioscience 49, 299–308.
- Smith P., Andrén O., Brussaard L., Dangerfield M., Ekschmitt K., Lavelle P., Tate K. (1998) Soil biota and global change at the ecosystem level: describing soil biota in mathematical models, Glob. Change Biol. 4, 773–784.
- Smith P., Smith J.U., Powlson D.S., McGill W.B., Arah J.R.M., Chertov O.G., Coleman K., Franko U., Frolking S., Jenkinson D.S., Jensen L.S., Kelly R.H., Klein-Gunnewiek H., Komarov A.S., Li C., Molina J.A.E., Mueller T., Parton W.J., Thornley J.H.M., Whitmore A.P. (1997) A comparison of the performance of nine soil organic matter models using datasets from seven long-term experiments, Geoderma 81, 153–225.
- Stark J.M., Firestone M.K. (1995) Mechanisms for soil moisture effects on activity of nitrifying bacteria, Appl. Environ. Microbiol. 61, 218–221.
- Stevenson F.J., Cole M.A. (1999) Cycles of soil. Carbon, nitrogen, phosphorous, sulfur, micronutrients, John Wiley & Sons, New York, USA.
- Thornley J.H.M. (1991) A transport-resistance model of forest growth and partitioning, Ann. Bot. 68, 211–226.
- Thornley J.H.M., Bergelson J., Parsons A.J. (1995) Complex dynamics in a carbon-nitrogen model of a grass-legume pasture, Ann. Bot. 75, 79–94.
- Van Gestel M., Merckx R., Vlassak K. (1993) Microbial biomass responses to soil drying and

rewetting: the fate of fast- and slow-growing microorganisms in soils from different climates, Soil Biol. Biochem. 25, 109– 123.

- Veldkamp E. (1994) Organic carbontTurnover in three tropical soils under pasture after deforestation, Soil Sci. Soc. Am. J. 58, 175–180.
- Verberne E.L., Hassink J., De Willigen P., Groot J.J.R., Van Veen J. (1990) Modelling organic matter dynamics in different soils, Neth. J. Agric. Sci. 38, 221–238.
- Verhoef H., Brussaard L. (1990) Decomposition and nitrogen mineralization in natural and agroecosystems: the contribution of soil animals, Biogeochemistry 11, 175–211.
- Vitousek P.M. (1984) Litterfall, nutrient cycling, and nutrient limitation in tropical forests, Ecology 65, 285–298.
- Vitousek P.M., Sanford R.L. (1986) Nutrient cycling in moist tropical forest, Annu. Rev. Ecol. Syst. 17, 137.
- Wattenbach M., Hattermann F., Weng R., Wechsung F., Krysanova V., Badeck F. (2005) A simplified approach to implement forest eco-hydrological properties in regional hydrological modelling, Ecol. Model. 187, 40–59.
- White R.E. (1997) Principles and practice of soil science, Blackwell Science, Oxford, UK.
- Wolf J., De Wit C., Van Keulen H. (1989) Modeling long-term crop response to fertilizer and soil nitrogen. I. Model description and application, Plant Soil 120, 11–22.
- Wolf J., Van Keulen H. (1989) Modeling long-term crop response to fertilizer and soil nitrogen. II. Comparison with field results, Plant Soil 120, 23–38.
- Xu L., Baldocchi D.D., Tang J. (2004) How soil moisture, rain pulses, and growth alter the response of ecosystem respiration to temperature, Glob. Biogeochem. Cycles 18.
- Xu T., Sonnenthal E., Spycher N., Pruess K. (2006) TOUGHR-EACT-A simulation program for non-isothermal multiphase reactive geochemical transport in variably saturated geologic media: applications to geothermal injectivity and CO<sub>2</sub> geological sequestration, Comput. Geosci. 32, 145–165.
- Young A., Young R. (2001) Soils in the Australian landscape, Oxford University Press, Victoria, Australia.
- Young T.P., Petersen D.A., Clary J.J. (2005) The ecology of restoration: historical links, emerging issues and unexplored realms, Ecol. Lett. 8, 662-673.
- Zelenev V.V., A.H.C. van Bruggen, Leffelaar P.A., Bloem J., Semenov A.M. (2006) Oscillating dynamics of bacterial populations and their predators in response to fresh organic matter added to soil: the simulation model "BACWAVE-WEB", Soil Biol. Biochem. 38, 1690–1711.
- Zheng D.W., Agren G.I., Bengtsson J. (1999) How do soil organisms affect total organic nitrogen storage and substrate nitrogen to carbon ratio in soils? A theoretical analysis, Oikos 86, 430–442.

# Greenhouse Gases and Ammonia Emissions from Organic Mixed Crop-Dairy Systems: A Critical Review of Mitigation Options

S.M. Novak and J.L. Fiorelli

#### Contents

1	Intro	duction 530		
2	Char Syste	acteristics of Organic Mixed Crop-Dairy ms		
3	Revie	ew of Mitigation Options Available for		
	Orga	nic Mixed Crop-Dairy Systems 533		
	3.1	Animal Production 534		
	3.2	Crop Production 542		
	3.3	Grasslands		
	3.4	Other Prospects		
4	Discu	ission		
5	Conc	lusion 550		
<b>References</b>				

S.M. Novak (⊠)

INRA, Unité Expérimentale Fourrages et Environnement, F-86600 Lusignan, France e-mail: sandra.novak@lusignan.inra.fr Abstract Dairy production systems represent a significant source of air pollutants such as greenhouse gases (GHG), that increase global warming, and ammonia (NH<sub>3</sub>), that leads to eutrophication and acidification of natural ecosystems. Greenhouse gases and ammonia are emitted both by conventional and organic dairy systems. Several studies have already been conducted to design practices that reduce greenhouse gas and ammonia emissions from dairy systems. However, those studies did not consider options specifically applied to organic farming, as well as the multiple trade-offs occurring between these air pollutants. This article reviews agricultural practices that mitigate greenhouse gas and ammonia emissions. Those practices can be applied to the most common organic dairy systems in northern Europe such as organic mixed crop-dairy systems. The following major points of mitigation options for animal production, crop production and grasslands are discussed. Animal production: the most promising options for reducing greenhouse gas emissions at the livestock management level involve either the improvement of animal production through dietary changes and genetic improvement or the reduction of the replacement rate. The control of the protein intake of animals is an effective means to reduce gaseous emissions of nitrogen, but it is difficult to implement in organic dairy farming systems. Considering the manure handling chain, mitigation options involve housing, storage and application. For housing, an increase in the amounts of straw used for bedding reduces NH<sub>3</sub> emissions, while the limitation of CH<sub>4</sub> emissions from deep litter is achieved by avoiding anaerobic conditions. During the storage of solid manure, composting could be an efficient mitigation

# **1** Introduction

There is currently a move towards more sustainable farming systems. One of the aims of sustainable agriculture is to establish environmentally-friendly production by limiting the adverse effects of agricultural activities on all the components of the environment. Regarding the atmosphere, agricultural activities may be a significant source of two major air pollutants: greenhouse gases (GHG), which contribute to global warming (IPCC, 2007), and ammonia (NH<sub>3</sub>), which can lead to eutrophication and acidification of natural ecosystems (Ferm, 1998). Livestock production has recently been questioned because of its environmental damage, particularly in terms of climate change and air pollution (Steinfeld et al., 2006). Figure 1 summarises the GHG and NH<sub>3</sub> fluxes of a mixed crop-dairy system.

Dairy production systems represent the largest agricultural source of the greenhouse gases methane  $(CH_4)$  and nitrous oxide  $(N_2O)$  in Europe (Weiske et al., 2006). In dairy husbandry, methane is mainly produced from enteric fermentation in the rumen of cows - approximately 80% - and to a lesser extent by cattle manures (Monteny et al., 2006). Agricultural soils are a sink for methane rather than a source, consumption rates of atmospheric methane in soils being very low in European soil and climatic conditions (Le Mer and Roger, 2001; Oenema et al., 2001). Keppler et al. (2006) recently reported CH<sub>4</sub> emissions by plant tissues under aerobic conditions, but these emissions seem not to significantly affect the CH<sub>4</sub> budget of grazed grasslands (Allard et al., 2007). N<sub>2</sub>O emissions are mainly derived from N inputs to agricultural land, i.e. from chemical fertilisers, manure, urine deposited by grazing animals, leguminous crops and crop residues, and from animal houses, i.e. from deep litter systems and solid manure heaps (Chadwick et al., 1999; Monteny et al., 2006; Petersen et al., 2006). In certain conditions, soils can also be a sink for N<sub>2</sub>O, but the factors regulating N<sub>2</sub>O consumption are not yet well understood (Chapuis-Lardy et al., 2007).

option, depending on its management. Addition of straw to solid manure was shown to reduce CH<sub>4</sub> and N<sub>2</sub>O emissions from the manure heaps. During the storage of liquid manure, emptying the slurry store before late spring is an efficient mitigation option to limit both CH<sub>4</sub> and NH<sub>3</sub> emissions. Addition of a wooden cover also reduces these emissions more efficiently than a natural surface crust alone, but may increase N2O emissions. Anaerobic digestion is the most promising way to reduce the overall greenhouse gas emissions from storage and land spreading, without increasing NH<sub>3</sub> emissions. At the application stage, NH<sub>3</sub> emissions may be reduced by spreading manure during the coolest part of the day, incorporating it quickly and in narrow bands. Crop production: the mitigation options for crop production focus on limiting CO<sub>2</sub> and N<sub>2</sub>O emissions. The introduction of perennial crops or temporary leys of longer duration are promising options to limit CO<sub>2</sub> emissions by storing carbon in plants or soils. Reduced tillage or no tillage as well as the incorporation of crop residues also favour carbon sequestration in soils, but these practices may enhance N<sub>2</sub>O emissions. Besides, the improvement of crop N-use efficiency through effective management of manure and slurry, by growing catch crops or by delaying the ploughing of leys, is of prime importance to reduce N<sub>2</sub>O emissions. Grassland: concerning grassland and grazing management, permanent conversion from arable to grassland provides high soil carbon sequestration while increasing or decreasing the livestock density seems not to be an appropriate mitigation option. From the study of the multiple interrelations between gases and between farm compartments, the following mitigation options are advised for organic mixed crop-dairy systems: (1) actions for increasing energy efficiency or fuel savings because they are beneficial in any case, (2) techniques improving efficiency of N management at field and farm levels because they affect not only N<sub>2</sub>O and NH<sub>3</sub> emissions, but also nitrate leaching, and (3) biogas production through anaerobic digestion of manure because it is a promising efficient method to mitigate greenhouse gas emissions, even if the profitability of this expensive investment needs to be carefully studied. Finally, the way the farmer implements the mitigation options, i.e. his practices, will be a determining factor in the reduction of greenhouse gas and NH<sub>3</sub> emissions.



Soil

Fig. 1 Greenhouse gas and ammonia fluxes in the main compartments of a mixed crop-dairy system (adapted from Soussana et al., 2004). Broken arrows indicate small fluxes. Emissions

from cattle excreta are indicated in the compartment "manures". Photos by D. Foissy, INRA Mirecourt

Carbon dioxide is another biogenic greenhouse gas playing a significant role in anthropogenic global warming. In dairy farming its major sources are soil and cattle. In soils, carbon dioxide is mainly released by microbial decay of plant litter and soil organic matter, and root respiration (Bahn et al., 2006). Cattle emit CO<sub>2</sub> through the respiration of organic carbon from ingested grass or fodder. However, this short-cycling carbon is generally seen as not relevant for the greenhouse effect, because it is assumed that carbon dioxide emissions have been fixed by plants through photosynthetic activity earlier in the farm cycle and thus make no net contribution to global warming (Schils et al., 2005). Soils can also act as a sink for  $CO_2$ ; the CO<sub>2</sub> fixed in plant biomass through photosynthesis can be stored in the soil as organic C by converting plant residues into soil organic matter after being returned to the soil. Root systems also make a significant contribution to soil C inputs through rhizodeposition (Rees et al., 2005). If the input of C into the soil is greater than its losses as CO<sub>2</sub>, C is stored in the soil.

In addition to their effect on global warming, agricultural activities and particularly livestock farming are the main source of atmospheric ammonia, around 50% of European ammonia emissions coming from cattle production (Ferm, 1998). In dairy farms,

losses of NH<sub>3</sub> occur during slurry application, housing, slurry storage, grazing, fertiliser application and from crops, in descending order of importance (Bussink and Oenema, 1998). After deposition on land, ammonia can increase acidification and nutrient-N enrichment of sensitive habitats (Sutton et al., 1993). In addition, NH<sub>3</sub> reacts with atmospheric acids to form ammonium  $(NH_4^+)$ -containing aerosol, which may both directly and indirectly affect light scattering and global radiative forcing (Sutton et al., 2001), and which is likely to threaten human health in Europe (WHO, 2004). Furthermore, ammonia is an indirect source of N<sub>2</sub>O: Mosier et al., (1998) assume that 1% of the atmospheric N deposited is converted into N<sub>2</sub>O, and Ferm (1998) estimates that 5% of the global  $N_2O$  emissions come from NH<sub>3</sub> oxidised in the atmosphere. N<sub>2</sub>O may also be indirectly emitted outside the farm gates from leached nitrate through denitrification.

To create more environmentally sustainable dairy systems regarding the air component, it is therefore necessary to limit the emissions of GHG, together with those of ammonia. Organic farming systems may be considered as the most developed form of sustainable farming systems, as their conception is based on objectives of environmental, social and economic sustainability. The fulfilment of the aim of environmental protection has already been studied by some authors assessing the effects of organic farming on various environmental parameters, including air quality (e.g. Stockdale et al., 2001; Shepherd et al., 2003), but to our knowledge, no literature study has reviewed agricultural practices that reduce emissions of GHG and NH<sub>3</sub> from organic dairy farming.

In this study we will analyse the mitigation options available for organic mixed crop-dairy systems, which are more common in Europe than specialised dairy or arable units (Stockdale et al., 2001). They are seen as systems having the potential to increase the efficiency of N cycling, thus reducing N losses to the environment (Ledgard, 2001). However, the combination of plant and animal production induces complex interrelations between the emissions of GHG and ammonia, which need to be studied before the implementation of mitigation options.

The mitigation options discussed in this article can also be applied beyond the scope of organic farming: they are relevant for all mixed crop-dairy systems which aim to be sustainable. Studies dealing with the mitigation of the three greenhouse gases  $N_2O$ ,  $CH_4$  and  $CO_2$ , as well as ammonia from dairy farming systems, are scarce (Brink et al., 2001; Wulf et al., 2002a), and consequences of mitigation options are mainly considered for individual sources or compartments of the farm and not at the whole-farm level. This article is a critical review of mitigation options accounting for direct  $N_2O$ ,  $CH_4$ ,  $CO_2$  and  $NH_3$  emissions as well as for indirect  $N_2O$  emissions through ammonia and nitrate losses and relevant for organic mixed crop-dairy systems in northern Europe.

## 2 Characteristics of Organic Mixed Crop-Dairy Systems

Organic systems aim to achieve a balanced relationship between soil, plants and animals (Vaarst et al., 2005). This is particularly the case in organic mixed crop-dairy systems, the crop production providing the animal feed and litter, while the animals provide the organic fertiliser through excreta (Fig. 2). Mixed cropdairy systems will be defined here as dairy systems producing forage crops for on-farm feeding, and characterised by the integration of livestock and arable crop production, with interchange of nutrients between crop and livestock production, described as the "highly integrated system" by Watson et al. (2005).

We will consider organic farming systems as defined by European Union legislation (EU, 1991, 1999, 2006). The 1804/99 European Union regulation (EU, 1999) describes in detail the allowed organic livestock



Fig. 2 Carbon and nitrogen flows in and out of a mixed crop-dairy system and between its main compartments. Gaseous losses from pre- and post-chains are outlined. Adapted from Olesen et al. (2006)

management in Europe. The main differences between organic and conventional systems concern housing and grazing conditions, animal nutrition, disease prevention and veterinary treatment. Zero grazing is not permitted and the maximum value of livestock density is limited to two livestock units per hectare for dairy cows. For livestock housing, at least half of the total floor must be solid and not of slatted or grid construction. Dry litter must be given in the rest area. Livestock must be fed on organically produced feedstuffs and 60% of the ruminant diet must come from forage. At least 50% of the feed shall come from the farm unit itself or if not, be produced in cooperation with other organic farms. The use of synthetic amino acids and growth promoters is forbidden, as well as the use of veterinary drugs in the absence of illness.

Regarding crop production and grasslands, the main differences between organic and conventional farming concern the ban on synthetic fertilisers and pesticides (EU, 1991). Organic systems are therefore characterised by low inputs of external nutrients and chemicals into the system. Nitrogen is brought from a legume "fertility-building" phase, through clover-based leys and arable grain legumes (Stopes et al., 2002), and also through lucerne (alfalfa)-based leys. The nitrogen fixed by legumes is made available to subsequent arable crops through mineralisation. Legume-based leys are utilised for grazing and fodder.

Grasslands are key components of organic mixed crop-dairy systems, by providing an important source of feed both through grazing and as hay and silage harvesting, and by contributing to building up the longterm fertility of the soil through organic matter deposition (Nicholas et al., 2004). Most organic systems have both temporary leys and permanent pastures (Nicholas et al., 2004).

Crop production is characterised by an increased diversity of cropping patterns in time and space compared with intensive conventional crop production systems (Stockdale et al., 2001). Intercrops or undersowing are often used to maintain soil cover yearround (Stolze et al., 2000).

The animal wastes provide organic fertiliser to croplands and grasslands, thus assuring the internal recycling of nutrients. As specified in Directive 91/676/EEC (EU, 1991), the total amount of manure applied on the farm may not exceed 170 kg N per year/ha of agricultural area used, which corresponds to 2 dairy cows/ha/yr as defined in Annex VII of the

2092/91 European Union regulation. The regulation also indicates that the number of livestock must be closely related to the area available in order to avoid problems of over-grazing and erosion, and to allow for the spreading of livestock manure, so that any adverse effect on the environment can be avoided.

The above-mentioned differences between the organic and conventional dairy systems imply that a lot of mitigation options proposed for conventional farming are not applicable to organic farms. For instance, products of synthesis which inhibit microorganisms or modify chemical reactions producing GHG such as methanogens in enteric fermentation or nitrification in soils are not allowed in organic farming. Besides, as organic systems and particularly mixed crop-dairy systems are extensive and use low external inputs, it does not leave much room for manoeuvre for lowering system nitrogen inputs or for reducing GHG emissions compared with intensive conventional systems which, for example, import energy-costly products such as pesticides or mineral nitrogen fertilisers.

Organic mixed crop-dairy systems can show some variations due to the environment in which they operate. The differences in livestock production will mainly consist of the land use (% of grasslands), the grassland management, the level of concentrates in the diet (between 0 and 40%), the dairy herd policy (size of the dairy herd, % of annual replacement, stocking density) and manure management. For instance, in the organic sector, Stolze et al. (2000) reported absolute livestock density of 1.6-1.8 livestock units (LU) per ha in the UK, but only 1.0 LU per ha on comparable organic farms in Germany. Some indications about the differences existing between organic dairy farming systems in Northern European countries can be found in Pflimlin and Kempf (2002), Weller (2002), Häring (2003), and Mosimann and Suter (2003). Crop production will mainly differ in crop rotation, soil tillage and fertilisation.

# 3 Review of Mitigation Options Available for Organic Mixed Crop-Dairy Systems

In general, mitigation measures may consist either of reducing or preventing the emissions of GHG and ammonia at source, or of favouring the storage of

**Table 1** Effect of slurry aeration on greenhouse gas and ammonia emissions during storage and after field application (from the results of Amon et al., 2006). Note that  $CO_2$  emissions mentioned here are not direct emissions, they correspond to a conversion of the energy requirement for the aeration process

	During stores				A fter engligeton			
	During	During storage				After applicatoli		
	$CH_4$	$CO_2$	$N_2O$	$NH_3$	$N_2O$	$NH_3$		
Slurry aeration	Ż	7	7	7	7	7		

carbon in plants or soils. The mitigation measures for dairy production systems can be either technical (e.g. manure application techniques), managementbased (e.g. changes in grazing/housing patterns) or system-orientated (e.g. shift from conventional to organic farming) Weiske et al., 2006).

Mitigation options show strong interrelations both between gases (more than one gas is affected either beneficially or adversely) and between farm compartments (the control of the emission of one compound may increase its emission at another stage of management). For instance, Amon et al. (2006) showed that intermittently aerating dairy cattle slurry had contrasting results on GHG and NH<sub>3</sub> emissions, and that slurry aeration affected the emissions not only during the storage period but also after slurry application (Table 1). It is therefore important to assess mitigation options for their impact upon the carbon and nitrogen cycles at the whole-farm level, by taking into account not only GHG and NH<sub>3</sub>, but also NO<sub>3</sub><sup>-</sup> leaching and C storage.

One must also consider that all the GHG do not affect global warming with the same intensity. Over the 100-year timescale, the Global Warming Potential from methane and nitrous oxide is estimated at 25 and 298, respectively, times that of CO<sub>2</sub> (Forster et al., 2007). A small increase in N<sub>2</sub>O emissions may thus counterbalance a large decrease in CH<sub>4</sub> emissions. Therefore, to be effective, mitigation options have to induce an overall reduction of GHG emissions in terms of CO<sub>2</sub> equivalents in addition to a reduction in NH<sub>3</sub> losses.

A large number of studies on reducing emissions from dairy systems have already been carried out, either for greenhouse gases (e.g. Velthof et al., 1998; Amon et al., 2001a; Clemens and Ahlgrimm, 2001; Monteny et al., 2006; Schils et al., 2006; Weiske et al., 2006) or for ammonia (e.g. Bussink and Oenema, 1998; Malgeryd, 1998; Webb et al., 2005). We will focus here on the mitigation strategies suitable for organic mixed crop-dairy systems in northern Europe, by considering their different compartments (Fig. 2), grouped together into three subsystems: i/ animal production, which includes livestock and manure management, ii/ crop production dealing with crop rotation, fertilisation and soil tillage, and iii/ grasslands, grazed (pastures) or cut (meadows). Due to the lack of bibliographic data, we will not consider the emissions generated by the feed and bedding stores themselves.

## 3.1 Animal Production

#### 3.1.1 Livestock Management

#### Feeding Strategies

Enteric fermentation in the cattle rumen is the major source of CH<sub>4</sub> emissions in dairy systems, but also the most difficult to reduce. The techniques currently being developed to reduce it by direct rumen manipulation (see e.g. Boadi et al., 2004; Martin et al., 2006) will not be discussed here as they are beyond the scope of organic farming. We will focus here on nutritional strategies consisting either of adding lipids to the diet, changing its fibre content or selective grazing. Linseed fatty acids supplemented at 6% were shown (Martin et al., 2008) to reduce dairy cow CH<sub>4</sub> emissions from 12 to 64% depending on the physical form of the lipid (crude linseed, extruded linseed or linseed oil). Furthermore, crude linseed increased milk yield, but not linseed oils. Linseed lipids are allowed as feed materials in organic dairy farming in the form of seeds or oil if the extraction process is physical (EU, 2006). Their promising results on CH<sub>4</sub> emissions need, however, to be confirmed by studies on more animals, taking into account the interaction with the nature of the basal diet and considering the effects of linseed on milk production.

Changing diet composition through the replacement of some of the grazing and roughage by concentrates (which have to represent less than 40% of the diet in organic dairy farming) increases animal productivity and may decrease the numbers of cows required to fill the annual milk quota (Lovett et al., 2006) and thus the CH<sub>4</sub> emissions per unit of milk. However, it goes

against the general aim of organic dairy farming to produce milk with a minimum of external inputs. Increasing the proportion of concentrate in the diet also has other direct and indirect effects on GHG and ammonia emissions. For instance, although supplementation of diets with concentrates often diminishes enteric methane emissions from cows, it may simultaneously enhance slurry methanogenesis because this is associated with extra amounts of undigested fibre which may be a substrate for slurry microbes (Hindrichsen et al., 2006). The fibre and the protein content of cattle diets were also shown to significantly influence the ammonia emission rates of manure (Kulling et al., 2003), as well as the plant availability of cattle slurry N and the amount of residual slurry N remaining in the soil after the first growing season (Sorensen et al., 2003). Moreover, replacing roughage by concentrates also contradicts the European environmental policy to promote extensive use of maintained grasslands, which store significant amounts of carbon in soil (Freibauer et al., 2004). Besides, the production and transport of concentrates generate GHG.

Because of its high starch content, the increased use of maize silage relative to dry matter intake was also suggested to reduce enteric CH<sub>4</sub> (Mills et al., 2001) and to improve animal performance. However, there is a need for animal studies that directly compare maize and other cereal silage with grass silage to quantify the reduction in CH<sub>4</sub> that might be achieved (Beauchemin et al., 2008). Furthermore, these inputs must be evaluated in terms of the net contribution to total GHG to evaluate if the substitution of maize or cereal silage for grass silage will result in a net reduction in GHG emissions on the farm scale.

Forage species also affect CH<sub>4</sub> production in ruminants (McAllister et al., 1996). Measurements of CH<sub>4</sub> production from grazing beef cows indicated a 25% reduction in CH<sub>4</sub> losses with alfalfa-grass pastures (7.1% of gross energy intake) compared with grass-only pastures (9.5% of gross energy intake) (McCaughey et al., 1999). The introduction of legumes into grazed grasslands would therefore limit CH<sub>4</sub> emissions. Legume species also seem to affect the CH<sub>4</sub> production from ruminants. For instance, condensed tannin-containing legumes (such as sulla) were shown to reduce the CH<sub>4</sub> emissions of dairy cows and other ruminants (Woodward et al., 2002; Tavendale et al., 2005).

The control of the protein intake of animals to meet their requirements more precisely is proposed to reduce gaseous emissions of nitrogen in grazed pastures, since excess N in the diet of cattle is to a large extent excreted in the urine, which in turn can influence ammonia and N<sub>2</sub>O emissions (Oenema et al., 2005; Watson et al., 2005; Kebreab et al., 2006). Controlling the N content of the diet is, however, not easy in organic dairy farming systems, since herbage plays an important role in the diet and because the proportion of legumes can vary greatly between swards through the seasons and between years. The N content of legumes is generally much higher than that required for optimum animal nutrition (Ledgard, 2001). However, as discussed above, CH<sub>4</sub> production (expressed relative to gross energy intake) from the rumen fermentation of legume forages is generally lower than the production from grass forage.

#### **Genetic Selection**

Genetic selection of cows based on an improvement of their ability to produce less methane or of their feed efficiency is still under validation (Boadi et al., 2004; Martin et al., 2006). The genetic selection of cows producing higher yield and thus less  $CH_4$  per unit of milk often leads to a decrease in animal fertility and health (especially with Holstein cows) and to an increase in the overall replacement rate (Lovett et al., 2006). This may paradoxically involve an increase in herd size, with heifers emitting GHG and NH<sub>3</sub>, which will increase the emissions from the whole farm.

#### Herd Characteristics

Options that increase lifetime efficiency or reduce the replacement rate are likely to reduce GHG emissions at the farm level, as shown in a modelling study from Weiske et al. (2006). This means that the decision to replace cows should not be based solely on economic considerations or fertility, but also on their capacity to produce milk for a long time. This could, for instance, be achieved by dairy cows resistant to illness and better adapted to the environment of their farm rather than high-yielding animals. Finally, reducing the milk production through the reduction of the number of milking cows certainly decreases the GHG and NH<sub>3</sub> emissions at the farm level, but this would mean a profound change in the farm economy.
		•		v	
Mitigation options fo	r livestock management	CH <sub>4</sub>	N <sub>2</sub> O	CO <sub>2</sub>	NH <sub>3</sub>
Feeding strategy	Adding linseed lipids to the diet	צ ?	-	-	_
	Increasing the proportion	<b>\</b> from animals	ס <b>ע</b> ? סר א	↗ (fossil energy)	⊿ or ы?
	of concentrate in the diet	<b> </b>	(from slurry <sup>a</sup> )	+ soil)	(from slurry <sup>a</sup> )
	Increasing the proportion of maize silage in the diet	<b>\</b> from animals	_	-	-
	Introducing legumes into grazed grasslands	R	<b>⊼</b> ?	צ ?	<b>7</b> ?
	Limiting excess N in the diet	<b>↗</b> ?	Ŕ	_	Ŕ
Genetic selection	Selecting cows with low enteric CH <sub>4</sub> production	צ ?	-	-	_
	Selecting high-yielding cows	<b>v</b> or <b>n</b> ?	-	_	<b>צ</b> <sup>b</sup> or <b>ז</b> ?
Herd characteristics	Reducing the replacement rate	צ ?	-	-	_
	Reducing the number of milking cows	R	R	И	Ŕ

Table 2 Effects on greenhouse gas and ammonia emissions of mitigation options reported for livestock management

<sup>a</sup> Depending on the N content of the concentrate compared to the roughage;

<sup>b</sup> Results from Lovett et al. (2006).

Caption:

 $\mathbf{v}$ : the mitigation option decreases the emissions

 $\mathbf{7}$ : the mitigation option increases the emissions

" $\checkmark$  or  $\checkmark$ ": both tendencies have been shown

- : no information was given on this compound

0: studies have shown that this option had no significant effect on this compound

?: the result needs to be confirmed by more studies.

In conclusion, Table 2 summarises the effects on GHG and  $NH_3$  emissions of the different mitigation options discussed for livestock management. In the current state of knowledge, two main mitigation strategies can be drawn from the above-mentioned options, based on either: 1/ a herd with a limited number of high-yielding dairy cows fed with an energy-rich ration but probably needing a high replacement rate; 2/ a herd with animals bred for hardiness, less productive but more robust and long-lived.

As methane production from enteric fermentation is rather hard to reduce, mitigation strategies from organic dairy farming have focused on the reduction of  $CH_4$  emissions from manure management.

#### 3.1.2 Manure Management

All parts of the manure handling chain, i.e. housing, storage and application, need to be considered, since intervening in one part affects losses in another. As mentioned in the introduction, the manure handling chain may be a source of GHG as well as ammonia. The organic European Union standards require that straw-based housing systems have to be used in organic livestock production. Manure management in organic dairy farming is thus based on the farmyard manure system, which may vary according to the amount of straw used. This amount will particularly determine whether all urine and faeces are retained by straw and stored as farmyard manure or if part of the urine is stored as liquid manure.

#### Housing

Ammonia losses from buildings are usually the second largest sources of  $NH_3$  after slurry application (Bussink and Oenema, 1998).  $NH_3$  emissions from the buildings may be reduced by increasing the amounts of straw used for bedding, with the advantage of no subsequent increase in  $NH_3$  losses during the storage or spreading of the manure, as all the ammoniacal nitrogen is immobilised in the straw (Webb et al., 2005).

Deep litter may result in significant emissions of  $N_2O$  and  $CH_4$ , depending on the rate of litter addition and mixing (Monteny et al., 2001; Oenema et al., 2005). Options to reduce  $CH_4$  emissions consist here of avoiding anaerobic conditions in the bedding.

### Manure Storage

Manure stores are the second largest source of methane emissions (after enteric fermentation) in European dairy farming (Sneath et al., 2006).  $CH_4$  emissions arise mainly from slurry stores, whereas farmyard manure stores are a significant source of N<sub>2</sub>O (Chadwick et al., 1999). Slurry and farmyard manure stored outside are also significant sources of NH<sub>3</sub> but they show great variations according to the temperature, the surface area, the duration of storage, and the occurrence of mechanical aeration (Bussink and Oenema, 1998).

The choice of a mitigation option at this stage of the manure handling chain will mainly depend on the nature of the effluent (liquid or solid manure).

# Mitigation Options During the Storage of Liquid Manure

**Overcoming the Effect of Storage Temperature** In northern European conditions, grazing is generally not possible during late autumn and in winter and cows are kept in buildings during this period. Slurry is therefore collected from cowsheds under cold conditions. Several studies have shown that CH<sub>4</sub> and NH<sub>3</sub> emission rates for slurry increased significantly with storage temperatures (Husted, 1994; Bussink and Oenema, 1998; Sommer et al., 2000; Clemens et al., 2006). For instance, Husted (1994) reported an increase of 61% in the overall methane emissions if the slurry was applied in September instead of June. Emptying the slurry store before the increase in air temperature (i.e. before late spring) would therefore limit the emissions of NH<sub>3</sub> and CH<sub>4</sub> that occur during the storage of liquid manure, as well as NH3 emissions resulting from field application which are also increased at higher air temperature (Sommer and Hutchings, 2001). However, other factors than air temperature must be considered, such as the load-bearing capacity of soils and the risk of nitrate leaching, which often limit the application of the stored slurry in autumn and winter. To overcome the effect of temperature when liquid manure is stored for a long time, some authors proposed cooling the manure tank (Clemens and Ahlgrimm, 2001; Monteny et al., 2001), but this requires fossil energy.

**Favouring the Formation of a Surface Crust** A natural crust generally forms at the slurry surface during storage as a result of evaporation, which promotes drying and binding of the particles if the slurry is not disturbed by mechanical mixing. The formation of this natural crust has been shown to greatly reduce NH<sub>3</sub> losses (by 50% in the study of Misselbrook et al. (2005) and by 80% in the study of Sommer et al. (1993)) and CH<sub>4</sub> emissions (by 38% in the study of Sommer et al. (2000) and by a factor of 12 in the study of Husted (1994)). Petersen et al. (2005) gave evidence for methanotrophic activity, i.e. methane oxidation, in surface crust materials. Measures to ensure crust formation are thus a cost-effective way of mitigating NH<sub>3</sub> and CH<sub>4</sub> emissions during the storage of slurry. They may, however, favour N<sub>2</sub>O emissions, which were shown to increase under warm conditions for slurry covered with a natural crust (Sommer et al., 2000).

The concentration and nature of the solids present in the slurry, which are influenced by the cattle diet and bedding material used, are important in determining crust formation, as well as environmental factors such as temperature, wind speed and rainfall (Misselbrook et al., 2005). These authors found that slurry dry matter content was the most important factor influencing crust formation, with no crust formation on slurries with a dry matter content below 1%. They also observed large differences in crust formation on slurries from grass silage-fed cattle bedded on long straw compared with corn silage-fed cattle bedded on chopped straw, but with no significant differences in NH<sub>3</sub> emission rates. Disturbance of the crust during the regular transfer of slurry from buildings could be minimised by inserting the nozzle of the liquid manure spreader below the surface crust, and homogenisation before emptying can be achieved by mixing with or without destroying the surface crust (Petersen et al., 2005).

**Covering Slurry Tanks** A wooden cover on slurry tanks has also been reported as an effective technique to reduce  $CH_4$  and  $NH_3$  emissions (Amon et al., 2006; Clemens et al., 2006), and  $N_2O$  emissions under cold conditions. Covering the tank may increase  $N_2O$  emissions in warm conditions, but as  $CH_4$  emissions during slurry storage contribute more to total GHG emissions than  $N_2O$  emissions, a lid has benefits as regards climate change. It may also shelter the natural surface crust from rain and help to keep it dry during winter. On the other hand, excluding rainwater from the slurry store may decrease the slurry water content, thus making its field application more difficult. Addition of a wooden cover reduces  $CH_4$  and  $NH_3$  emissions from slurry more than a natural surface crust alone. Covers made with other materials, such a chopped straw or wood, or expanded clay, have been shown to reduce NH<sub>3</sub> losses but not overall GHG (and particularly  $CH_4$ ) emissions (Amon et al., 2006; Clemens et al., 2006; Guarino et al., 2006).

**Performing Anaerobic Digestion** Anaerobic digestion of manure has been proposed by many authors (e.g. Clemens et al., 2006; Monteny et al., 2006; Weiske et al., 2006) as a measure to abate  $CH_4$  emissions from manure storage. Anaerobic digestion is a natural process whereby bacteria existing in oxygen-free environments decompose organic matter, resulting in a biogas (a mixture of  $CH_4$ ,  $CO_2$  and some trace gases) and a sludge that is stable and nearly odourless (Kebreab et al., 2006). Methane produced by digesters can be captured and burned as fuel. Anaerobic digestion in a covered gas-tight plant thus offers the benefits of reducing trace gas emissions and substituting fossil fuels by renewable energy (biogas).

Few studies have been done on the GHG and NH<sub>3</sub> emissions of slurry after its anaerobic digestion. During winter storage in pilot-scale slurry tanks, Clemens et al. (2006) reported significantly lower CH<sub>4</sub> emissions from digested dairy cattle slurry than from the same untreated slurry (emissions of N<sub>2</sub>O and NH<sub>3</sub> were similar for the two slurries). These authors emphasise the importance of the digestion's duration (hydraulic retention time), which must be sufficiently long for a complete degradation of fermentable organic matter, so as to exploit the potential for gas production without increasing CH<sub>4</sub> emissions during subsequent storage.

However, even with a sufficiently long hydraulic retention time, CH<sub>4</sub> is still produced after anaerobic digestion. Clemens et al. (2006) therefore recommend including all potentially gas-producing compounds within biogas plants for complete collection of CH<sub>4</sub> and optimum environmental and economic benefit. It is also important to prevent uncontrolled losses of methane from biogas plants (which occur, e.g., by mean of small leakages) which can considerably alter the environmental balance of a biogas production system (Borjesson and Berglund, 2006).

After field application, Clemens et al. (2006) found no significant differences in GHG emissions between untreated and digested cattle slurry, whereas Petersen (1999) reported lower N<sub>2</sub>O emissions from digested slurry compared with untreated slurry. NH<sub>3</sub> losses after field application of digested slurry have been reported to be similar to (Rubaek et al., 1996; Clemens et al., 2006; Wulf et al., 2002b), lower than (Rubaek et al., 1996) or higher than (Amon et al., 2006) those from raw slurry. These contrasting results are probably attributable to the effects of anaerobic digestion on both chemical and physical properties of the slurry. Anaerobic digestion reduces manure carbon and dry matter content by about 50% but generally results in an increase in  $NH_4^+$  content and pH (Wulf et al., 2002b; Amon et al., 2006). The reduced viscosity of digested slurry improves its infiltration, thus limiting the NH<sub>3</sub> volatilisation, whereas its higher  $NH_4^+$  content and pH promote the NH<sub>3</sub> losses. NH<sub>3</sub> emissions from digested slurry can therefore be reduced if the increased potential loss due to chemical factors is compensated for by its faster infiltration (Wulf et al., 2002b). Techniques to reduce NH<sub>3</sub> emissions from field application will be discussed later.

In France, Ademe et al. (2006) estimated 200–250 k euros as the cost of a biogas installation producing 30 kW of electricity. In organic dairy farming, two other limitations must be taken into account: 1/ slurry is only collected at one time of the year (i.e. during housing) and 2/ co-digestion with other waste products to increase gas production is not feasible, particularly because they should also come from organic farming unless the digested product is not used for an organic farm. The economical feasibility of on-farm power/heat generation with biogas will depend to a great extent on the energy price and on subsidies.

**Performing a Mechanical Separation** Mechanical separation is another technique used to treat slurry, resulting in a liquid fraction with low dry matter content reduced by 40–45%, and a solid fraction that can be stored in heaps. Separation is performed with a screw sieve separator and uses little energy. The separated slurry has a lower viscosity and flows more easily through band-spreading hoses. The study of Amon et al. (2006) indicates that slurry separation reduces  $CH_4$  emissions, but is likely to result in an increase in  $N_2O$  and  $NH_3$  emissions during composting of the solid fraction.

Lowering the pH of Slurry Lowering the pH of slurry with additives such as lactic acid has also been proposed to avoid the production of methane, this acid being authorised in organic farming not intentionally for that purpose but for the cleaning and the disinfection of livestock buildings and installations. Lowering the pH of the slurry with lactic acid can indeed reduce both CH<sub>4</sub> and N<sub>2</sub>O emissions, but Berg et al. (2006) showed that this reduction is effective only if the pH is below 6.0, and that a lower pH would be necessary to reduce ammonia emissions. This technique seems therefore not to be practically possible.

Aerating the Slurry As oxygen is a strong inhibitor of methane production and an easily available product, aeration seems an attractive method to limit CH<sub>4</sub> emissions. Aeration was indeed shown to reduce CH<sub>4</sub> emissions of slurry during its storage (Martinez et al., 2003; Amon et al., 2006), but N<sub>2</sub>O and NH<sub>3</sub> emissions may be greatly increased when the slurry is aerobically treated (Béline et al., 1999; Amon et al., 2006). N<sub>2</sub>O may also be emitted by the aerated slurry during its subsequent storage (Béline et al., 1999). Besides, slurry aeration results in energy consumption. Nevertheless, laboratory studies showed that the aeration strategy (intermittent or continuous, length of the anoxic period) would play an important role in the emissions of N<sub>2</sub>O (Béline et al., 2002). However, these results need to be confirmed by experiments on farmscale units.

**Comparison of Treatments** Comparing the effect of different treatments of dairy cattle slurry (wooden or straw cover, mechanical separation, anaerobic digestion, aeration), Amon et al. (2006) found that anaerobic digestion reduced the overall GHG emissions from storage and land spreading the most, and had hardly any influence on NH<sub>3</sub> emissions from these processes (Fig. 3). This study has the asset of having been conducted in pilot-scale tanks whose conditions are nearer to field conditions than studies with laboratory reactors.

### Mitigation Options During the Storage of Solid Manure

The use of straw for bedding in organic dairy housing systems produces solid manure, which can be either composted or stored outdoors stockpiled.

**Composting Solid Manure** Organic systems encourage the composting of solid manure, i.e. aerobic decomposition at temperatures of around 60°C, which

#### Emissions (%w versus untreated slurry)



**Fig. 3** Relative ammonia and greenhouse gas emissions from storage and land spreading for different treatments of dairy cattle slurry compared with untreated slurry with a wooden cover (from Amon et al., 2006)

offers the advantages of producing a more stable, uniform product, free of weeds and toxins and easier to spread (Mustin, 1987; Peigne and Girardin, 2004). However, the heat generated and the regular aeration by turning of the heap during the early stages of composting favour ammonia losses which are generally greater than for stockpiled farmyard manure (Amon et al., 2001b; Gibbs et al., 2002; Peigne and Girardin, 2004). Although NH<sub>3</sub> losses are large during the composting phase, they are low after subsequent soil application, because the remaining N is mainly in organically bound forms and thus concentrations of inorganic N are very low (Sommer and Hutchings, 2001; Amon et al., 2001b; McNeill et al., 2005).

Mechanical turning of composted manure generally results in lower N2O and CH4 emissions than for anaerobically stacked farmyard manure (Amon et al., 2001b; Peigne and Girardin, 2004). Considering the whole management system (housing, storage and spreading), Amon et al. (2001b) reported GHG  $(CH_4 + N_2O)$  emissions 25% lower from a composting system than from an anaerobic stacking system. Significant amounts of CO<sub>2</sub> may, however, be emitted during the composting process, but it is not a net source of CO<sub>2</sub> along the recycling chain of agricultural wastes, since well-composted material contains less easily decomposable carbon compounds than nondecomposed material (Peigne and Girardin, 2004). Composted manures being less degradable than fresh manures, their application permits greater carbon storage in the soil (Kirchmann and Bernal, 1997).

GHG and ammonia emissions during composting are greatly influenced by its management (turning method and frequency, duration of the composting operations) and by the physical and chemical characteristics of the raw material (C/N ratio) (Peigne and Girardin, 2004). A high porosity of the initial material provides sufficient aeration, which is essential for a good composting process and for limiting N<sub>2</sub>O and CH<sub>4</sub> emissions. An increase in the C/N ratio significantly reduces N losses during storage and composting of livestock solid manure (Sommer, 2001). If the density of the farmyard manure is high (low porosity) or the C/N ratio is low (<20), straw may be added to the farmyard manure to improve its composting. Furthermore, the number and frequency of turnings can be reduced if the raw material has good characteristics (Peigne and Girardin, 2004), thus limiting NH<sub>3</sub> losses.

Considering indirect  $N_2O$  emissions, a concrete platform with water recovery is the best system to prevent nitrate losses by leaching or runoff from the compost heap to surface or groundwater (Peigne and Girardin, 2004). Covering the pile with a waterproof but air-permeable cover is another way to avoid contamination by runoff (Peigne and Girardin, 2004). However, nitrate leaching losses from compost with a high C/N ratio are generally low (Sommer, 2001).

Compacting and Covering Manure Heaps Compacting and covering manure heaps has the potential to reduce emissions of both NH3 and N2O when the manure contains relatively high ammonium-N contents (Chadwick, 2005). Compaction may, however, increase CH<sub>4</sub> emissions (Chadwick, 2005; Oenema et al., 2005). Covering manure heaps may also not always be effective, because the mineralisation rate increases with temperature, and so the effect of increased mineralisation may exceed the effect of reduced exchange rates of NH<sub>3</sub> from the manure with the atmosphere (Bussink and Oenema, 1998). Addition of straw to solid manure is another way of reducing NH<sub>3</sub> emissions by improving the C/N ratio (Clemens and Ahlgrimm, 2001), and it was also shown to reduce CH<sub>4</sub> and N<sub>2</sub>O emissions (Yamulki, 2006).

Reductions in  $NH_3$  emissions at the manure storage stage will result in increased manure total ammoniacal nitrogen content and therefore in potentially larger losses following land spreading (Misselbrook et al., 2005). It is therefore important that suitable application or incorporation methods are used to minimise losses.

#### **Application Techniques**

Slurry application is the major source of ammonia emissions in dairy farming systems (Bussink and Oenema, 1998) and it is to a lesser extent a direct source of N<sub>2</sub>O (Oenema et al., 2005). Significant NH<sub>3</sub> and N<sub>2</sub>O emissions also occur following the application of farmyard manure (Webb et al., 2004). In addition to being a direct source of N<sub>2</sub>O, manure spreading may be an indirect source of this gas through ammonia losses and nitrate leaching. In general, the reduction of NH<sub>3</sub> emissions is more effective at the spreading phase than from buildings (Webb et al., 2005).

#### Spreading Manure During the Coolest Part of the Day

A "simple" mitigation method is to limit the application of manure to when conditions do not favour  $NH_3$ volatilisation, e.g. during the coolest part of the day. Sommer and Olesen (2000) have calculated that avoiding applications during times of the day with a high potential for  $NH_3$  losses could reduce the total emission of  $NH_3$  from applied slurry by half. However, the efficiency of this technique depends on the farmer's flexibility in the choice of application date and time (Sommer and Hutchings, 2001).

### Incorporating Manure

Several techniques of manure spreading on land have been shown to decrease NH<sub>3</sub> losses (Sommer and Hutchings, 2001). Among them, rapid incorporation of manures into arable land by ploughing is the most costeffective (Webb et al., 2005). Incorporation should be as soon after application as possible, especially after slurry application, as loss rates are high in the first hours after application (Sommer and Hutchings, 2001). For slurry, the use of disk or tine cultivators may be as effective as ploughing (Webb et al., 2005). Incorporating manure was found to increase (Ferm et al., 1999) or decrease N<sub>2</sub>O emissions by soils (Webb et al., 2004). It is therefore not possible to generalise the effects of manure incorporation into the soil on N<sub>2</sub>O emission, most probably because the denitrification process in soils depends on many factors (see e.g. Oenema et al., 2005; Mathieu et al., 2006).

### Spreading the Slurry in Bands

The application of slurry in narrow bands (i.e. band spreading), rather than over the entire soil or crop

surface, by trailing hose or trailing shoe was also shown to be an effective technique to reduce  $NH_3$ losses after liquid manure spreading (Webb et al., 2005). The effectiveness of band spreading increases if the slurry is placed below the canopy of a welldeveloped crop (Webb et al., 2005). Trail hoses or trail shoes have the advantage of limiting N<sub>2</sub>O emissions in comparison with slurry injection, which reduces  $NH_3$ emissions reliably but may greatly enhance N<sub>2</sub>O emissions (Flessa and Beese, 2000). Injection of slurry also causes higher fuel consumption (Wulf et al., 2002a). Trail hoses are, in general, more effective on arable land than on grassland and when used with dilute than with more viscous slurries, whereas trail shoes are designed more for grasslands (Webb et al., 2005).

Investigating the effect of different application techniques on greenhouse gas and ammonia emissions from co-fermented slurry, Wulf et al. (2002a) found that the most effective methods to limit both GHG and ammonia emissions were trail hose application with immediate shallow incorporation for arable land, and trail shoe application for grassland.

One must also consider the indirect effects on GHG emissions of an improved application of manure in the field (Weiske et al., 2006). For instance, as  $NH_3$  losses from the manure are reduced, there are less indirect emissions of  $N_2O$  due to deposition of  $NH_3$  outside the field. But as more nitrogen is effectively applied to soil, nitrate leaching is likely to increase and hence the derived  $N_2O$  emissions. This can, however, be counterbalanced by increased crop yields resulting from the increased amount of nitrogen available for the crops.

The effects of timing applications to match crop demand are discussed later in the "crop production" section.

#### Solid Versus Liquid Manure

Depending on the amount of straw used for bedding, the manure handling chain will manage strawbased manure only, both solid (straw-based) and liquid (slurry-based) manures or even liquid manure only. We report here the results of studies or reviews comparing the emissions of GHG or NH<sub>3</sub> of straw-manure-based systems versus slurry-based ones. From different experimental studies, Monteny et al. (2006) concluded that animal housing and manure stores of straw-based systems (deep litter) result in greater N<sub>2</sub>O emissions than the more anaerobic slurry-based systems. On the other hand, several studies reported substantially lower  $N_2O$  emissions following the application of solid manure compared with the application of liquid manure (Gregorich et al., 2005).

Concerning CH<sub>4</sub>, Stolze et al. (2000) estimated that lower potential CH<sub>4</sub> emissions are likely for strawbased manure because stable manure has a significantly lower metabolic factor for methane than liquid manure. Conversely, Monteny et al. (2001) hypothesised that deep litter systems may emit larger quantities of CH4 than slurry-based systems, due to the increased temperature in the deep litter bed and (partially) anoxic conditions resulting from compaction by the animals. On the other hand, Hansen et al. (2002) estimated that CH<sub>4</sub> emissions from deep litter were similar to those of slurry-based housing systems. The application of solid manure generally results in higher C storage than slurry (Foereid and Hogh-Jensen, 2004; Ceotto et al., 2006). Concerning NH<sub>3</sub>, increased use of straw for bedding is likely to reduce NH<sub>3</sub> emissions from housing (Bussink and Oenema, 1998; Stockdale et al., 2001).

These sometimes conflicting results show the difficulty in drawing conclusions about the magnitude of GHG and NH<sub>3</sub> emissions from straw-based manures compared with slurry-based ones when the whole farming system is considered. It is all the more difficult to give a generalised conclusion, since GHG and NH<sub>3</sub> emissions closely depend on the techniques and practices used for livestock housing, manure storage and manure application. Besides, an efficient comparison of solid and liquid manure systems needs to take into account the emissions generated by the production of the straw. Finally, at the whole-farm level, a change in the manure management system has implications on the balance between grassland and arable areas, these latter determining the production of straw used for the bedding.

In conclusion, Table 3 summarises the effects on GHG and  $NH_3$  emissions of the different mitigation options discussed for the manure handling chain. An overall reduction of GHG and  $NH_3$  emissions at the manure handling chain requires that mitigation options are implemented on all its parts, from housing to application. For housing, the reduction of  $NH_3$  and  $CH_4$  emissions may be achieved by a judicious management of the bedding, whereas among the several techniques available at the storage level, anaerobic

Mitigation options for manure management		CH <sub>4</sub>	N <sub>2</sub> O	CO <sub>2</sub>	NH <sub>3</sub>	
Housing	Increasing the amounts of straw used for bedding	-	-	И	-	
	Avoiding anaerobic conditions in the bedding	-	-	-	R	
Storage	Emptying the slurry store before the increase in air temperature	-	-	Ŕ	R	
	Cooling the manure tank	-	<b> </b>	и И	К	
	Favouring the formation of a surface crust	<b>↗</b> ?	_	Ŕ	Ы	
	Covering slurry tanks	<ul><li>▶ in cold conditions,</li><li>↗ in warm conditions</li></ul>	-	Ŕ	R	
	Performing the anaerobic digestion of the slurry	at field application: $\checkmark$ or 0	_	at field application: 0, オ or ⊾	Ы	
	Performing a mechanical separation	<b>л</b> ?	0	<b>Л</b> ?	R	
	Lowering the pH of slurry	Ŕ	-	-	К	
	Aerating the slurry	<b>⊅</b> ?	<b> </b>	7	К	
	Composting solid manure	Я	<ul><li>↗ during composting</li><li>↘ at application</li></ul>	<ul><li>↗ during composting</li><li>↘ at application</li></ul>	R	
	Compacting and covering manure heaps	Ŕ	-	צ ?	⊿?	
	Adding straw to solid manure	צ ?	-	Ŕ	צ?	
Application techniques	Spreading manure during the coolest part of the day	_	-	И	-	
	Incorporating manure (rapidly)	א or א	_	Ŕ	-	
	Spreading the slurry bands with trail hoses or trail shoes	<ul> <li>↘ in comparison with slurry injection</li> </ul>	<b>\</b> (fossil energy)	и	-	
Solid versus liquid manure		0?, <b>↗</b> ? or <b>↘</b> ?	<ul> <li>at housing and storage</li> <li>at application</li> </ul>	(higher carbon storage)	R	

Table 3 Effects on greenhouse gas and ammonia emissions of mitigation options reported for manure management

digestion seems the most promising to reduce the overall GHG emissions from storage and land spreading, without increasing NH<sub>3</sub> emissions. At application, band spreading effectively reduces emissions, all the more since the manure is applied to arable land by trail hose and rapidly incorporated.

### 3.2 Crop Production

For the crop production sub-system, the mitigation options focus on limiting  $CO_2$  and  $N_2O$  emissions, which are the main gases produced at this level. The

limitation of net  $CO_2$  emissions from cropping systems may be achieved either by increasing the C storage in soils or plants, or by slowing the return of stored C into the atmosphere via mineralisation (or fire). For instance, measures such as the incorporation of straw aim at enhancing the input of carbon to the soil, while measures such as reduced tillage aim at reducing the output of carbon from the soil. Mitigation options dealing with N<sub>2</sub>O emissions are mainly based on an improvement of the N-use efficiency of crops, because N<sub>2</sub>O is generated by soil microorganisms largely from surplus mineral N, which may also lead to nitrate leaching (Syväsalo et al., 2006; Kuikman et al., 2004; Smith et al., 2008). To a lesser extent, mitigation options aim to improve soil structure to limit anaerobic sites favouring  $N_2O$  emissions (Ball et al., 1999). We will present the main mitigation options reducing  $N_2O$  and  $CO_2$  emissions of the crop production system by considering in turn crop rotation, fertilisation and soil tillage.

### 3.2.1 Crop Rotation

Increasing diversity in crop rotation is generally associated with a positive effect on C sequestration (West and Post, 2002), especially when legumes are introduced (INRA, 2002; Gregorich et al., 2005). Some crop species favour C storage in soils, such as crops with large deep root systems and perennial crops, which are generally regarded as allocating more C below-ground (Rees et al., 2005; Smith et al., 2008). On the other hand, silage maize, which is characterised by a low return of crop residues, is thought to decrease the C stock of soils (INRA, 2002). Perennial crops also have a beneficial effect on N2O emissions during winter (Gregorich et al., 2005) and on nitrate leaching losses (Watson et al., 2005), which have generally been shown to be smaller from perennial than from annual crops following manure application.

Organic cropping systems are generally characterised by temporary leys which alternate with arable crops. Temporary leys have a potential for soil C storage intermediate between crops and permanent grassland (Soussana et al., 2004). Prolonging the lifespan of temporary leys permits one not only to maintain their C storage for a longer time, but also to increase their storage capacity (INRA, 2002). Part of the additional carbon stored in the soil during the ley phase is released when it is ploughed up (Soussana et al., 2004). Ploughing of the ley results in a rapid mineralisation of soil organic matter and thus in large CO<sub>2</sub> emissions. The release of large quantities of N from mineralisation of grass/clover residues may also favour nitrate leaching (Berntsen et al., 2006) and may significantly contribute to the total N<sub>2</sub>O emission from cultivated soils (Flessa et al., 2002; Vellinga et al., 2004).

Nitrate losses following grassland cultivation may be controlled by good management practices, such as delaying ploughing until late winter or spring for spring crops (Watson et al., 2005). If spring incorporation of the ley is not possible, it is advised to establish the following crop early in autumn to maximise N uptake during autumn (Berry et al., 2002) or to use efficient catch crops after ploughing. For instance, Eriksen et al. (2004) showed a considerable decrease in nitrate leaching when replacing the rotation "winter wheat - bare soil", following the ploughing of the ley, by spring oats and ryegrass catch crops in the two winters. More generally, cover crops or "catch crops" may be planted when the period between two main crops is rather long, with the advantage of reducing nitrate leaching after harvest of the main crop in autumn and increasing the total duration of photosynthesis (INRA, 2002). The immobilised soil nitrogen is subsequently made available after incorporation by mineralisation. Therefore, careful attention to the timing and method of incorporation of the cover crop, to synchronise mineralisation with periods of high crop demand (Hu et al., 1997), is very important. However, there are still uncertainties on how decomposition and mineralisation of cover crop residues that are incorporated into soil will affect nitrate leaching over the long term (McNeill et al., 2005). For enhancing the C storage from cover crops it is preferable to incorporate them with reduced tillage.

### 3.2.2 Fertilisation

Mitigation options concerning fertilisation are mainly based on an improvement of the N efficiently used by crops to reduce mineral N accumulation in the soil, which may generate both N<sub>2</sub>O emissions and nitrate leaching.

#### Improving N-Use Efficiency

Practices improving N-use efficiency consist of synchronising N inputs with crop growth and crop uptake, such as, for instance, adjusting application rates with crop needs or avoiding time delays between N application and plant N uptake (improved timing). However, these mitigation measures reviewed by Monteny et al. (2006) were mainly devised for farming systems using mineral fertilisers and not for organic systems where the supply of soil mineral N is the sum of biological fixation from legumes, and direct inputs of mineral N from atmospheric deposits and manure, plus mineralisation from soil organic matter and other organic materials (crop residues, manure).

N fixed by legumes represents the major source of N inputs into organic cropping systems but its quantification is very uncertain (Nicholas et al., 2004), because its magnitude is regulated by changes in soil inorganic N and competition from associated grasses (Ledgard and Steele, 1992). For instance, the percentage of legumes in a ley decreases with increasing manure application (Hansen, 1996). This feedback mechanism has, however, the advantage of limiting the N inputs to legume/grass soil and consequently regulating the potential for N losses to watercourses or air (Syväsalo et al., 2006). Besides, according to Rochette and Janzen (2005), much of the increase in soil N<sub>2</sub>O emissions in legume crops would be attributable to the N released from root exudates during the growing season and from decomposition of crop residues after harvest, rather than from biological N fixation per se. Carter and Ambus (2006) also found that recently fixed N released via easily-degradable clover residues was a minor source of N<sub>2</sub>O in a grass-clover grassland.

The soil mineral nitrogen coming from the mineralisation of soil organic matter is also hard to predict, since the actual amount and timing of the mineralisation is influenced by a number of factors including soil moisture, aeration and temperature, the nature and accessibility of the organic matter, previous fertilisation, intensity and timing of cultivation, and cropping patterns (Stockdale and Rees, 1995; Rasmussen et al., 1998; Berry et al., 2002). Matching nitrogen supply from the mineralisation of soil organic matter with crop demand is thus a difficult exercise.

In organic farming, improvements in N-use efficiency are particularly related to an effective use of manures as nutrient resources. This effective management of manure and slurry must take into account that nutrients in farmyard manure are available at a slower rate for plant uptake than from slurry (Watson et al., 2005) and that uncomposted manure contains more readily available N than composted manure (Berry et al., 2002). As manures vary widely in composition, depending on feed composition, quality and quantity of bedding material, length of storage and storage conditions (Shepherd, 2000; Watson et al., 2002), manure analysis may help to predict its available N. For instance, N content and the C/N ratio of animal wastes are highly correlated with the N mineralisation rate (Morvan et al., 2006) and they may be used to predict it. The use of agronomic models to provide decision support systems or appropriate management guidelines for organic N fertilisation is promising but still needs further development (David et al., 2005; Morvan et al., 2006). These models could help to improve manure and slurry use; for instance, by determining the type of manure most beneficial to the different crops of the rotation over longer periods than a single crop or growing season to ensure both short-term productivity and long-term sustainability (Watson et al., 2005). Future improvements in N utilisation within organic systems could also be made by breeding crops with traits that improve the capture of available N and the efficiency with which is it converted to yield (Berry et al., 2002).

Improved timing of manure application may also help to prevent nitrate leaching, which generally occurs when large amounts of nitrate in the soil profile coincide with a period of high drainage (Di and Cameron, 2002). In areas with light-textured soils, manures applied in spring pose a smaller risk for leaching than when they are applied to bare soil in autumn (McNeill et al., 2005). Splitting the annual N application rates was also shown to reduce nitrate leaching losses for pasture systems, by improving the synchrony between the pasture N demand and supply (Di and Cameron, 2002). Composting the manure may also be effective at reducing nitrate leaching losses, which were shown to be lower from the application of composted manure compared with its raw materials (Basso and Ritchie, 2005).

More generally, strategies reducing N leaching will also limit N2O emissions. Besides, modelling studies showed that improving farm efficiency of N management would reduce overall GHG emissions from conventional as well as from organic dairy farms (Schils et al., 2006; Olesen et al., 2006). However, N budgets for organic farms are less reliable than those for conventional farms, particularly because N fixation represents the predominant N input source for organic farms (see e.g. Steinshamn et al., 2004) and its estimation remains highly uncertain even with the use of empirical models. Besides, Olesen et al. (2006) indicated that the N efficiency is strongly affected by the ratio of crop to animal products in the farm output. Further investigations are therefore necessary to confirm this relationship between farm N efficiency and GHG emissions for organic dairy farms.

### Timing the Effluent Application with Soil Wetness

Another strategy, proposed by de Klein and Eckard (2008) to limit N<sub>2</sub>O emissions resulting from

fertilisation, is the timing of effluent application in relation to soil wetness.  $N_2O$  emissions were indeed shown to be higher when the slurry was applied to wet soil compared with drier soil (Saggar et al., 2004).

### Sequestering C Through an Improved Fertilisation

Manure management also affects soil organic carbon sequestration in different ways, the two main ones being biomass production and N availability. Firstly, fertilisation increases the primary productivity of crops and therefore the quantities of above- and belowground residues added into soil in unharvested plant parts (Rees et al., 2005). However, the decomposition rates of the crop residues (mineralised into CO<sub>2</sub>) could also be increased for high fertilisation inputs (Rees et al., 2005). Secondly, soil N availability is one of the main factors affecting humus formation in agricultural soils (Christopher and Lal, 2007). Furthermore, manures are a source of C per se, and the fate of their C in soil (mineralised versus humified) depends particularly on the biochemical characteristics of their organic matter (e.g. Morvan et al., 2006), and therefore on the cattle's diet and on the straw characteristics. As stated by Christopher and Lal (2007): "the challenge of sequestering C in agricultural soils is to increase the concentration of humus while producing good crop yields and maintaining low concentrations of N in soil solution and discharging waters". The hypothesis put forward by Smith et al. (2000) that the application of farm manures to arable soils can increase the soil carbon stock to a greater extent than application to grasslands was questioned by Soussana et al. (2004).

### 3.2.3 Soil Tillage

Ploughing is an efficient practice for weed management in organic cropping systems. However, soil tillage stimulates the decomposition of soil organic matter, particularly by disrupting soil aggregates (Balesdent et al., 2000).

### Limiting Soil Tillage

Limiting soil disturbance by reduced (shallow) tillage or no tillage therefore decreases the decomposition rate of soil organic carbon (Smith et al., 2000; Rees et al., 2005). Depending on the study concerned, reduced tillage was as effective as no tillage at storing C in soil (INRA, 2002) or it did not induce significant change in soil organic carbon (Kern and Johnson, 1993). Effectiveness of conversion to conservation tillage on soil organic carbon sequestration also depends on soil and climatic factors such as texture, temperature and water availability (Lal, 2004a; Rees et al., 2005). For instance, light-textured and well-drained soils in moist and cool climates would sequester more soil organic carbon than clayey and poorly-drained soils (Lal, 2004b). Reduced tillage would also be more effective where relatively high soil carbon contents occur simultaneously with relatively high decomposition rates (Vleeshouwers and Verhagen, 2002).

The permanence of the tillage practices is another determining factor in soil carbon sequestration, because soil carbon sequestered in arable soils is impermanent and is lost more rapidly than it accumulates (Freibauer et al., 2004). Agricultural soils that are tilled every few years may contain more carbon than the same soils cultivated every year (Smith et al., 1997), but even a single ploughing can drastically accentuate emissions of  $CO_2$  from soil, because of an increase in the rate of mineralisation (Reicosky et al., 1999).

One must also take into account that the absence of tillage may lead to higher  $N_2O$  emissions than conventional tillage, due to larger soil aggregates, low gas diffusivity and greater water retention near the soil surface making the soil less aerobic (Ball et al., 1999; Smith et al., 2001). The results of studies comparing  $N_2O$  emissions from conventional versus reduced tillage are, however, conflicting (see e.g. Holland, 2004; Smith and Conen, 2004), like those of studies comparing manure incorporation into the soil (as previously reported).

Moreover, the differences in  $N_2O$  fluxes between the two tillage systems are likely to change over time. Six et al. (2004) showed increased  $N_2O$  fluxes in the first 10 years of adoption of no-tillage, but lower  $N_2O$  fluxes in humid climates in no-till than conventional till after 20 years. These results can be explained by the transient lower crop yields and the greater soil water content generally observed in recently established no-till systems (Six et al., 2004), or by an improvement of the soil structure with time (Holland, 2004). Conversely, methane uptake was reported to be significantly increased under no-tillage (Six et al., 2004). Finally, considering the three GHG fluxes together, Six et al. (2004) found that no-tillage leads to an increase in net global warming potential during the first 10 years, but to a strong decrease after 20 years of adoption in humid climates.

West and Post (2002) and West and Six (2007) found greater C sequestration for a decrease in tillage intensity as compared with an increase in rotation complexity, but they emphasised that increase in soil C following a change in rotation complexity may occur over a slightly longer period of time. The sequestration of soil organic carbon resulting from a change in tillage may also depend on the crop species (Wright et al., 2007).

None of these results take into account the direct machinery energy consumption. The cultivation of soils by ploughing is the most fuel-consuming process in the production of arable crops (Holland, 2004). Conservation tillage has the additional benefit of saving fuel and therefore of limiting  $CO_2$  emissions (see e.g. Filipovic et al., 2006), even if systems based upon conservation tillage may require additional operations such as the creation of a stale seedbed (Holland, 2004).

#### Avoiding Soil Compaction

Soil compaction by tractor wheels and tillage machinery may increase  $N_2O$  emission (Ball et al., 1999) by favouring the development of anaerobic zones within some of the soil structural units as a result of the loss of air-filled macropores (McNeill et al., 2005). Therefore, avoiding compaction by traffic and tillage may help to reduce  $N_2O$  emissions, especially since compaction has a great effect on yield (Hansen, 1996; Stockdale et al., 2001) and thus on C storage and nitrate leaching.

#### Incorporating Crop Residues

Incorporation of crop residues tends to increase soil organic matter, thus storing C in the soil, but N-rich crop residues may stimulate rapid denitrification and associated nitrous oxide emissions, even in coarse-textured soils (Velthof et al., 2002). From a N<sub>2</sub>O mitigation point of view, incorporating residues with low N content is better than a homogeneous mixing of N-rich materials into the soil (Ambus et al.,

2001). Incorporation of N-rich, low C/N ratio residues leads to rapid mineralisation and a large rise in soil mineral N, while residues low in N, such as cereal straw, can lead to net immobilisation of N in the short to medium term (Watson et al., 2002). On livestock farms, however, straw is more likely to be used for bedding. Moreover, based on a review of published Canadian studies, Gregorich et al. (2005) concluded that ploughing manure or crop stubble into the soil in the autumn led to higher levels of N2O production  $(2.41 \text{ kg N-N}_2\text{O} \text{ ha}^{-1} \text{ year}^{-1})$  than if residues were left on the soil surface (1.19 kg N-N<sub>2</sub>O ha<sup>-1</sup> year<sup>-1</sup>). Mineralisation of N from crop residues may also result in nitrate leaching, depending on the quantity of N in the plant material and its C/N ratio (Berry et al., 2002), on the synchrony between N release and plant demand (Eriksen et al., 2006), and on soil texture.

The N content of crop residues also influences soil C sequestration. Incorporation of crop species producing residues with high C/N ratios (such as wheat) was shown to favour C sequestration compared with residues with low C/N ratios, which stimulated decomposition of native soil organic matter (Wright et al., 2007). Nevertheless, even low C/N ratio materials are beneficial for C sequestration after their incorporation (Rees et al., 2005).

In any case, crop residues should not be burned to avoid emissions of aerosols (Hays et al., 2005) and of CO<sub>2</sub>, resulting in C losses for the system (Lal, 2004b).

In conclusion, Table 4 summarises the effects on GHG and  $NH_3$  emissions of the different mitigation options discussed for crop production. The main mitigation options concern the introduction of perennial crops or the longer duration of temporary leys, a reduction of the tillage or the improvement of crop N-use efficiency through effective management of manure and slurry, by growing catch crops or by delaying the ploughing of leys. The long-term effects of these mitigation options are still uncertain because of the difficulty of predicting and controlling the C and N dynamics of organic matter in soils.

### 3.3 Grasslands

In this section we will discuss conversion from arable to grassland and review mitigation options applicable to grazing management.

Mitigation opti	ions for crop production	$CH_4$	N <sub>2</sub> O	CO <sub>2</sub>	NH <sub>3</sub>
Crop rotation	Increasing diversity in crop rotation	_	_	Ŕ	_
	Introducing perennial crops	_	R	<i>L</i>	_
	Prolonging the lifespan of temporary leys	-	– <b>オ</b> after ploughing	▶ <b>A</b> after ploughing	-
	Cultivating catch crops	-	▲ at short term  A ? at long term	■ at short term <b>&gt;</b> ? at long term	-
Genetic selection	Breeding crops improving N use efficiency	-	<b>צ</b> ?	-	-
Fertilisation	Synchronizing N inputs with crop uptake	-	Ŕ	_	-
	Timing effluent application with soil wetness	-	Ŕ	-	-
	Improving the fertilisation	-	<b>↗</b> ?	<b>≥</b> ? or <b>↗</b> ?	-
Soil tillage	Reducing tillage	⊿?	<b>↗</b> ? or <b>`&gt;</b> ?	И	_
	Avoiding soil compaction	_	R	<i>L</i>	_
	Incorporating crop residues	-	<b>↗</b> ?	Ŕ	-

Table 4 Effects on greenhouse gas and ammonia emissions of mitigation options at the crop production stage

### 3.3.1 Conversion from Arable to Grassland

Several studies comparing farmland management options in Europe have indicated that conversion from arable to grassland (and to a lesser extent from temporary grassland to permanent grassland) provides high soil C sequestration potential (INRA, 2002; Smith, 2004). The results of Ammann et al. (2007) suggest, however, that the conversion of arable land to managed grassland has a positive effect on the carbon balance during the first 3 years only if the system receives nitrogen inputs. A moderate increase in N supply to nutrient-poor permanent grasslands has also been shown in long-term surveys to increase grassland topsoil organic carbon stocks, except for nutrientpoor grasslands developed on organic soils (Soussana et al., 2004). To be effective, the conversion from arable to grassland should remain permanent, because the ploughing under of grassland results in large CO<sub>2</sub> and N<sub>2</sub>O emissions (Vellinga et al., 2004). Also, carbon losses are much faster after returning grassland to arable use than the build-up of soil carbon when establishing grassland (Soussana et al., 2004).

Until recently, the general assumption was that soil carbon fluxes were in balance in permanent pasture, i.e. in the absence of changes in environmental factors and in land use and management, an equilibrium value was reached for all soil organic C pools (e.g. Freibauer et al., 2004; Lovett et al., 2006). However, the recent results of Soussana et al. (2007) do not confirm this concept of carbon sink saturation for

permanent semi-natural grasslands, which displayed net carbon storage. Their study indicated that extensively managed (i.e. semi-natural) but N-rich grasslands may store more carbon than highly intensive (i.e. newly sown grass-clover mixtures) grasslands.

Conversion of annual fodder or cereal crops into temporary or permanent grasslands or conversion of temporary grassland into permanent grassland means an increasing proportion of grass in the diet of animals, which often corresponds with an extensification of the livestock production at the farm level.

### 3.3.2 Grazing Management

#### Livestock Density

We will consider here the possible mitigation options induced by the management of the livestock density with a constant herd size, and not the limitation of emissions resulting from a reduction of the number of animals on the farm. Under continuous grazing systems, the effect of stocking rate on methane production per animal is not consistent. For instance, McCaughey et al. (1997) observed a decrease in absolute CH<sub>4</sub> emissions (g d<sup>-1</sup>) and CH<sub>4</sub> yield (CH<sub>4</sub> energy loss as a percentage of gross energy intake, % of GEI) with increasing stocking rates of steers (1.1 versus 2.2 steer ha<sup>-1</sup>), while Pinares-Patino et al. (2007) found only a reduction in the CH<sub>4</sub> yields (% GEI) and not in the absolute CH<sub>4</sub> emissions per

Mitigation options for grasslands		CH <sub>4</sub>	$N_2O$	$CO_2$	NH3
Land use	Permanent conversion from arable				
	to grassland (to a less extent from	_	-	L.	_
	temporary to permanent grassland)				
Grazing	Increasing livestock density	▶? or 0?	↗?	↗?	-
management	Selective grazing associated with grass maturity	צ ?	-	-	-

**Table 5** Effects on greenhouse gas and ammonia emissions of mitigation options for grasslands

animal with increasing stocking rates of heifers (1.1. versus 2.2 LU ha<sup>-1</sup>). When pastures were rotationally grazed, stocking rates had no effect on CH<sub>4</sub> production (McCaughey et al., 1997).

Increasing livestock density seems not to be an efficient mitigation option, since the nitrogen surplus increases with increasing stocking density (Stockdale et al., 2001; Dalgaard et al., 2002). It is therefore likely to enhance the potential for N<sub>2</sub>O losses from urine and dung excreted during grazing, especially since the increased trampling of grazing animals causes soil compaction and thus an increase in the number of anaerobic sites on the soil, which in turn may favour N<sub>2</sub>O emissions (Oenema et al., 1997). The surplus nitrogen may also be lost by nitrate leaching: Eriksen et al. (1999) observed a tendency towards increased nitrate leaching losses with increasing stocking density. Moreover, modelling results from Soussana et al. (2004) indicated that the magnitude of the C sink of grazed grasslands declined with the mean annual stocking rate. On the other hand, a drastic reduction in livestock density is also not recommended, since Allard et al. (2007) showed that grassland management methods that abruptly reduced grazing pressure (1.2 to 0.6 LU ha<sup>-1</sup> year<sup>-1</sup>) and fertiliser input (80 to 0 kg N ha<sup>-1</sup> year<sup>-1</sup>) gradually reduced the C storage potential of the grassland.

### Grazing System

There is limited information with regard to the effects of rotational grazing versus continuous grazing on CH<sub>4</sub> production. For steers, McCaughey et al. (1997) observed that at low stocking rates (1.1 steer ha<sup>-1</sup>), CH<sub>4</sub> production (L ha<sup>-1</sup> d<sup>-1</sup>) was 9% lower for rotational grazing than continuous grazing.

Selective grazing which is associated with grass maturity is likely to limit  $CH_4$  emissions. Pinares-Patino et al. (2003) showed that the grazing of grass at early heading induced a decrease of 10% in

CH<sub>4</sub> emissions compared with the grazing of the same grass at a later stage of maturity when digestibility was lower. Nevertheless, CH<sub>4</sub> emissions are generally more determined by feed intake rather than feed digestibility (McCaughey et al., 1999; Pinares-Patino et al., 2007), an increase in feeding level inducing lower CH<sub>4</sub> losses as a fraction of gross energy intake (Boadi et al., 2004).

As EU 1804/99 regulation specifies that the rearing systems for herbivores are to be based on maximum use of "pasturage", we will not discuss mitigation options affecting the length of the grazing period. This maximum use of grasslands should, however, be managed in order to prevent adverse effects on soil (compaction) or vegetation (through overgrazing), which affect  $N_2O$  and  $CH_4$  emissions (soil compaction) or nitrate leaching (scarce vegetation), as discussed previously.

In conclusion, Table 5 summarises the effects on GHG and  $NH_3$  emissions of the different mitigation options discussed for grasslands. The main mitigation option consists of converting arable to grassland, especially permanent grasslands which allow C sequestration. Concerning grazing management, a higher livestock density may lead to a decrease in  $CH_4$  emissions by the grazing animals, but to an increase in  $N_2O$  emissions from their excreta and to a decrease in the C stored by the grasslands. In the same way, selective grazing associated with lower stocking density allows a decrease in  $CH_4$  emissions through an increasing digestibility of intake compared with a more complete use of grass.

### 3.4 Other Prospects

In this paragraph we give some leads on other types of mitigation options that may apply to organic mixed crop-dairy systems. Their effects on GHG and NH<sub>3</sub> emissions are summarised in Table 6. Hedgerows

Table 6 Effects on green-           baues gas and ammonia	Mitigation options CH <sub>4</sub>				CO <sub>2</sub>	NH <sub>3</sub>
emissions of mitigation	Arable field	Introducing hedgerows or tree belts	-	-	R	R
options for other prospects	Farm scale	Increasing energy efficiency	-	-	ĸ	-
	Crop	Producing plant oil	-	-	R	-

or tree belts may be introduced in arable field margins to store carbon, and to recapture part of the NH<sub>3</sub> emitted into the air by the farm (Theobald et al., 2002). Increasing the energy efficiency is also a way to reduce CO<sub>2</sub> emissions. Diesel savings may be achieved by technological solutions (e.g. better mechanical efficiency of tractors, use of wind or solar energy instead of fossil fuel) or by improved farm machinery management (e.g. optimised field operations and work planning, see e.g. Dyer and Desjardins 2003; Couvreur, 2006). In the context of organic mixed crop-dairy systems, the production of biomass for energy (bioenergy crops) is not feasible on a large scale. However, the production of small quantities of plant oil that can be used as fuel and which also provide oil-seed cakes for livestock feeding (see e.g. Brunschwig and Lamy, 2006) is worth studying, though these crops should not be grown in preference to pastures.

### 4 Discussion

As discussed at the beginning of this article, to judge the appropriateness of mitigation options, it is essential to assess their impact on the carbon and nitrogen cycles at the whole-farm level. However, as highlighted in this review, it is far from easy because each mitigation option involves trade-offs between gases, and because there are often interactions and feedbacks among mitigation options. The choice of a set of mitigation options will therefore rely on the careful assessment of the balance between their beneficial and adverse effects. The establishment of this balance should consider not only the overall emissions of GHG and NH<sub>3</sub> resulting from the implementation of the chosen mitigation options (including fuel carbon costs for farm operations), but also their environmental and agronomic side effects. GHG and NH<sub>3</sub> mitigation measures often have accompanying benefits on other environmental components than global warming. For instance, C storage in soils has several unintended beneficial effects on soil fertility and water quality (see e.g. Lal, 2004a) and hedgerows used to store C in vegetation have many other valuable functions (see e.g. Marshall and Moonen, 2002) such as windbreaks or habitat for wild plants and animals. In addition to the consideration of environmental side effects, it is important to consider the effects of mitigation measures on animal welfare, which is another objective of organic farming. Animal welfare may be affected in several ways: e.g. through the feeding, housing conditions (type of bedding), presence of shelters (trees) on the pasture, length of the grazing period, etc. We will not consider in this article the economic and social aspects of the implementation of mitigation options (see e.g. Smith et al., 2007).

In this article we have examined the effects of mitigation options at the farm level. But to draw conclusions about their positive effects on global change, it seems important to consider also the emissions generated by the production and transport of materials needed to implement the mitigation measures (e.g. purchased concentrates, biogas plant). Besides, mitigation options also have to be devised for the transport of dairy farm products (exported from the farm). To reduce fuel consumption, an off-farm strategy consists, for instance, of moving to a more regionally-based economy, in which the producer and consumer are brought closer together (Johnson et al., 2007).

Another factor to take into consideration is the period of time needed to establish the balance between beneficial and adverse effects of mitigation options. GHG (especially N<sub>2</sub>O) and NH<sub>3</sub> emissions vary greatly over time, particularly because of the influence of meteorological conditions on physicochemical and biological reactions governing them. The effectiveness of a mitigation measure will thus show strong year-to-year variations. Furthermore, the delay between the implementation of a mitigation measure and the reduction of the emissions will greatly vary between mitigation options. For instance, an improvement in the method of applying manure will have an immediate effect on NH<sub>3</sub> emissions, whereas a change in land management or in soil tillage affecting the soil properties (and thus soil carbon sequestration and

N<sub>2</sub>O emissions) will probably involve a time scale of several years. Also, the kinetics of carbon accumulation following change in land use or grassland management are non-linear, i.e. they are more rapid during the early years after adopting a practice which enhances accumulation (INRA, 2002). The balance of a mitigation measure may thus change with time. For instance, as discussed above, Six et al. (2004) found that zero tillage leads to an increase in net global warming potential during the first 10 years, but to a strong decrease after 20 years of adoption in humid climates.

The soil and climatic context of each mixed cropdairy system will be important for determining the most effective set of mitigation options. The soil characteristics, particularly its hydromorphy, will play an important role in the N<sub>2</sub>O emissions and in soil organic matter mineralisation; it will also affect the best time for manure application or soil tillage (trafficability), and even the balance between grassland and arable areas. This balance may also determine the amount of straw in the manure but it is not clear whether slurry or farmyard manure will be more beneficial for global warming.

Decision support tools would be useful to assess the balance of different sets of mitigation options by taking into consideration trade-offs, interaction and feed-back among practices for different time scales and at the farm level, and by evaluating their impact upon environmental and agronomic components. Simulation models already exist on the dairy farm scale (reviewed by Schils et al., 2007; Table 7) but they still need improvements to accurately evaluate mitigation options from organic dairy farms. In particular, GHG emissions resulting from legume incorporation and from deep litter mats need further research to be accurately modelled. Simulation modelling, combining biophysical and decisional models, would also help farmers with their management decisions at both strategic and tactical levels.

Besides, the assessment of mitigation options (and the validation of models) is only possible if accurate measurements of the GHG and NH<sub>3</sub> emissions or of the carbon storage are available. Measuring greenhouse gas emissions for the different farm components, however, presents serious difficulties (see e.g. the review of McGinn, 2006), particularly because emissions are characterised by a high spatial and temporal variability. Measurements are necessary, not only to assess the efficiency of mitigation options but also to identify all the major sources of GHG emissions of mixed crop-dairy systems, some of which may currently be underestimated, such as, for instance, ditches or fresh heaps of maize silage (Hensen et al., 2006).

# **5** Conclusion

The most promising mitigation options are firstly measures that increase energy efficiency or fuel savings, because they are beneficial in any case. Secondly, techniques improving efficiency of N management at field and farm levels should also be promoted because they affect not only N<sub>2</sub>O and NH<sub>3</sub> emissions, but also nitrate leaching. A better use of N at the farm level is, however, confronted by difficulties in organic mixed crop-dairy systems, either to optimise the total N intake of the livestock or to improve the N-use efficiency of crop and grass production. For instance, the management of the N content of the diet has to cope with the varying and unpredictable proportion of legumes in pastures, and the mineral N content of

	DairyWise	FarmGHG	SIMS <sub>DAIRY</sub>	FarmSim
Authors	Schils et al. (2006)	Olesen et al. (2006)	Del Prado et al. (2006)	Saggar et al. (2004)
Model type	Empirical	Empirical	Semi-mechanistic	Semi-mechanistic
CH <sub>4</sub> and N <sub>2</sub> O emissions	х	Х	Х	Х
CO <sub>2</sub> emissions	х	х		х
C sequestration				Х
NH <sub>3</sub> and NO <sub>3</sub> emissions	х	Х	Х	Х
Pre-chain emissions	х	Х		Х
Animal welfare			Х	
Biodiversity			Х	
Soil quality			Х	

Table 7 General characteristics of whole-farm greenhouse gas models (adapted from Schils et al., 2007)

the soil is difficult to control through application of manures and incorporation of crop residues. Thirdly, biogas production through anaerobic digestion of manure seems a promising and efficient way to mitigate GHG emissions, but the profitability of this expensive investment needs to be studied in the local context before its implementation. Last but not least, the way the farmer will implement the mitigation options, i.e. his practices, will be a determining factor in the reduction of GHG and NH<sub>3</sub> emissions. Some techniques such as composting may or may not be considered as a mitigation option, depending on the farmer's practices.

Options aiming to reduce GHG and NH<sub>3</sub> emissions result in numerous trade-offs between air pollutant emissions and nutrient flows that may occur in all the components of the farming system, and that may involve time scales of decades or even more. Before its implementation, a mitigation option should therefore be assessed in the context of the whole farming system or even larger by including pre- and post-chains, at least on the time scale of the crop rotation and if possible, by considering other issues than global warming, e.g. water quality, soil fertility, animal welfare, biodiversity, etc. The interactions and feedback that can take place among a set of mitigation options also have to be considered. The determination of the most effective set of mitigation options would thus benefit from the development of decision support tools based on dynamic models of the C and N cycles of the whole farming system.

### References

- Ademe, Aile, Solagro, Trame (2006) La méthanisation à la ferme, Paris, p. 16.
- Allard V., Soussana J.-F., Falcimagne R., Berbigier P., Bonnefond J.M., Ceschia E., D'hour P., Henault C., Laville P., Martin C., Pinares-Patino C. (2007) The role of grazing management for the net biome productivity and greenhouse gas budget (CO<sub>2</sub>, N<sub>2</sub>O and CH<sub>4</sub>) of semi-natural grassland, Agr. Ecosyst. Environ. 121, 47–58.
- Ambus P., Jensen E.S., Robertson G.P. (2001) Nitrous oxide and N-leaching losses from agricultural soil: Influence of crop residue particle size, quality and placement, Phyton-Ann. REI Bot. 41, 7–15.
- Ammann C., Flechard C.R., Leifeld J., Neftel A., Fuhrer J. (2007) The carbon budget of newly established temperate grassland depends on management intensity, Agr. Ecosyst. Environ. 121, 5–20.

- Amon B., Amon T., Alt C., Moitzi G., Boxberger J. (2001a) Nitrous oxide emissions from cattle production systems and mitigation options, Phyton-Ann. REI Bot. 41, 17–28.
- Amon B., Amon T., Boxberger J., Alt C. (2001b) Emissions of NH<sub>3</sub>, N<sub>2</sub>O and CH<sub>4</sub> from dairy cows housed in a farmyard manure tying stall (housing, manure storage, manure spreading), Nutr. Cycl. Agroecosyst. 60, 103–113.
- Amon B., Kryvoruchko V., Amon T., Zechmeister-Boltenstern S. (2006) Methane, nitrous oxide and ammonia emissions during storage and after application of dairy cattle slurry and influence of slurry treatment, Agr. Ecosyst. Environ. 112, 153–162.
- Bahn M., Knapp M., Garajova Z., Pfahringer N., Cernusca A. (2006) Root respiration in temperate mountain grasslands differing in land use, Global Change Biol. 12, 995–1006.
- Balesdent J., Chenu C., Balabane M. (2000) Relationship of soil organic matter dynamics to physical protection and tillage, Soil Till. Res. 53, 215–230.
- Ball B.C., Scott A., Parker J.P. (1999) Field N<sub>2</sub>O, CO<sub>2</sub> and CH<sub>4</sub> fluxes in relation to tillage, compaction and soil quality in Scotland, Soil Till. Res. 53, 29–39.
- Basso B., Ritchie J.T. (2005) Impact of compost, manure and inorganic fertilizer on nitrate leaching and yield for a 6-year maize-alfalfa rotation in Michigan, Agr. Ecosyst. Environ. 108, 329–341.
- Beauchemin K.A., Kreuzer M., O'Mara F., McAllister T.A. (2008) Nutritional management for enteric methane abatement: a review, Aust. J. Exp. Agr. 48, 21–27.
- Béline F., Martinez J., Chadwick D., Guiziou F., Coste C.M. (1999) Factors affecting nitrogen transformations and related nitrous oxide emissions from aerobically treated piggery slurry, J. Agr. Eng. Res. 73, 235–243.
- Béline F., Martinez J. (2002) Nitrogen transformations during biological aerobic treatment of pig slurry: effect of intermittent aeration on nitrous oxide emissions, Bioresource Technol. 83, 225–228.
- Berg W., Brunsch R., Pazsiczki I. (2006) Greenhouse gas emissions from covered slurry compared with uncovered during storage, Agr. Ecosyst. Environ. 112, 129–134.
- Berntsen J., Grant R., Olesen J.E., Kristensen I.S., Vinther F.P., Molgaard J.P., Petersen B.M. (2006) Nitrogen cycling in organic farming systems with rotational grass-clover and arable crops, Soil Use Manage. 22, 197–208.
- Berry P.M., Sylvester-Bradley R., Philipps L., Hatch D.J., Cuttle S.P., Rayns F.W., Gosling P. (2002) Is the productivity of organic farms restricted by the supply of available nitrogen? Soil Use Manage. 18, 248–255.
- Boadi D., Benchaar C., Chiquette J., Masse D. (2004) Mitigation strategies to reduce enteric methane emissions from dairy cows: Update review, Can. J. Anim. Sci. 84, 319–335.
- Borjesson P., Berglund M. (2006) Environmental systems analysis of biogas systems - Part 1: Fuel-cycle emissions, Biomass Bioenerg. 30, 469–485.
- Brink C., Kroeze C., Klimont Z. (2001) Ammonia abatement and its impact on emissions of nitrous oxide and methane - Part 2: application for Europe, Atmos. Environ. 35, 6313–6325.
- Brunschwig P., Lamy J.M. (2006) Production à la ferme d'huile végétale et de tourteaux : possibilités et conséquences, Fourrages 187, 329–342.

- Bussink D.W., Oenema O. (1998) Ammonia volatilization from dairy farming systems in temperate areas: a review, Nutr. Cycl. Agroecosyst. 51, 19–33.
- Carter M.S., Ambus P. (2006) Biologically fixed N<sub>2</sub> as a source for N<sub>2</sub>O production in a grass-clover mixture, measured by N-15(2), Nutr. Cycl. Agroecosyst. 74, 13–26.
- Ceotto E., Borrelli L., Marchetti R., Tomasoni C. (2006) Effect of integrated forage rotation and manure management systems on soil carbon storage, in: Ramiran International Conference: Technology for recycling of manure and organic residues in a whole-farm perspective, DIAS, pp. 29–32.
- Chadwick D.R. (2005) Emissions of ammonia, nitrous oxide and methane from cattle manure heaps: effect of compaction and covering, Atmos. Environ. 39, 787–799.
- Chadwick D.R., Sneath R.W., Phillips V.R., Pain B.F. (1999) A UK inventory of nitrous oxide emissions from farmed livestock, Atmos. Environ. 33, 3345–3354.
- Chapuis-Lardy L., Wrage N., Metay A., Chottes J.L., Bernouxs M. (2007) Soils, a sink for N<sub>2</sub>O? A review, Global Change Biol. 13, 1–17.
- Christopher S.F., Lal R. (2007) Nitrogen management affects carbon sequestration in North American cropland soils, Crit. Rev. Plant Sci. 26, 45–64.
- Clemens J., Ahlgrimm H.J. (2001) Greenhouse gases from animal husbandry: mitigation options, Nutr. Cycl. Agroecosyst. 60, 287–300.
- Clemens J., Trimborn M., Weiland P., Amon B. (2006) Mitigation of greenhouse gas emissions by anaerobic digestion of cattle slurry, Agr. Ecosyst. Environ. 112, 171–177.
- Couvreur J.-P. (2006) Quelques leviers pour maîtriser la consommation d'énergie et optimiser la mécanisation de l'exploitation, Fourrages 187, 301–310.
- Dalgaard T., Heidmann T., Mogensen L. (2002) Potential N-losses in three scenarios for conversion to organic farming in a local area of Denmark, Eur. J. Agron. 16, 207–217.
- David C., Jeuffroy M.H., Laurent F., Mangin M., Meynard J.M. (2005) The assessment of Azodyn-Org model for managing nitrogen fertilization of organic winter wheat, Eur. J. Agron. 23, 225–242.
- de Klein C.A.M., Eckard R.J. (2008) Targeted technologies for nitrous oxide abatement from animal agriculture, Aust. J. Exp. Agr. 48, 14–20.
- del Prado A., Merino P., Estavillo J.M., Pinto M., González-Murua C. (2006) Nutrient Cycling in Agroecosystems, 74, 229–243. DOI: 10.1007/S10705-006-9001-6
- Di H.J., Cameron K.C. (2002) Nitrate leaching in temperate agroecosystems: sources, factors and mitigating strategies, Nutr. Cycl. Agroecosyst. 64, 237–256.
- Dyer J.A., Desjardins R.L. (2003) Simulated farm fieldwork, energy consumption and related greenhouse gas emissions in Canada, Biosyst. Eng. 85, 503–513.
- Eriksen J., Askegaard M., Kristensen K. (1999) Nitrate leaching in an organic dairy/crop rotation as affected by organic manure type, livestock density and crop, Soil Use Manage. 15, 176–182.
- Eriksen J., Pedersen L., Jorgensen J.R. (2006) Nitrate leaching and bread-making quality of spring wheat following cultivation of different grasslands, Agr. Ecosyst. Environ. 116, 165–175.

- Eriksen J., Vinther F.P., Soegaard K. (2004) Nitrate leaching and N-2-fixation in grasslands of different composition, age and management, J. Agr. Sci. 142, 141–151.
- EU (1991) Council Regulation No. 2092/1991 of 24 June 1991, Off. J. Eur. Communities L198, 1–5.
- EU (1999) Council Regulation No. 1804/1999 of 19 July 1999, Off. J. Eur. Communities L222, 1–28.
- EU (2006) Council Regulation No. 1791/2006 of 20 November 2006, Off. J. Eur. Communities L363, 1–80.
- Ferm M. (1998) Atmospheric ammonia and ammonium transport in Europe and critical loads: a review, Nutr. Cycl. Agroecosyst. 51, 5–17.
- Ferm M., Kasimir-Klemedtsson A., Weslien P., Klemedtsson L. (1999) Emission of NH<sub>3</sub> and N<sub>2</sub>O after spreading of pig slurry by broadcasting or band spreading, Soil Use Manage. 15, 27–33.
- Filipovic D., Kosutic S., Gospodaric Z., Zimmer R., Banaj D. (2006) The possibilities of fuel savings and the reduction of CO<sub>2</sub> emissions in the soil tillage in Croatia, Agr. Ecosyst. Environ. 115, 290–294.
- Flessa H., Beese F. (2000) Laboratory estimates of trace gas emissions following surface application and injection of cattle slurry, J. Environ. Qual. 29, 262–268.
- Flessa H., Ruser R., Dorsch P., Kamp T., Jimenez M.A., Munch J.C., Beese F. (2002) Integrated evaluation of greenhouse gas emissions (CO<sub>2</sub>, CH<sub>4</sub>, N<sub>2</sub>O) from two farming systems in southern Germany, Agr. Ecosyst. Environ. 91, 175–189.
- Foereid B., Hogh-Jensen H. (2004) Carbon sequestration potential of organic agriculture in northern Europe - a modelling approach, Nutr. Cycl. Agroecosyst. 68, 13–24.
- Fontaine S., Bardoux G., Abbadie L., Mariotti A. (2004) Carbon input to soil may decrease soil carbon content, Ecol. Lett. 7, 314–320.
- Forster P., Ramaswamy V., Artaxo P., Berntsen T., Betts R., Fahey D.W., Haywood J., Lean J., Lowe D.C., Myhre G., Nganga J., Prinn R., Raga G., Schulz M., Van Dorland R. (2007) Changes in Atmospheric Constituents and in Radiative Forcing, in: Solomon S., Qin D., Manning M., Chen Z., Marquis M., Averyt K.B., Tignor M., Miller H.L. (Eds.), Climate Change 2007: The Physical Science Basis, Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Freibauer A., Rounsevell M.D.A., Smith P., Verhagen J. (2004) Carbon sequestration in the agricultural soils of Europe, Geoderma 122, 1–23.
- Gibbs P.A., Parkinson R.J., Misselbrook T.H., Burchett S. (2002) Environmental impacts of cattle manure composting, in: Microbiology of Composting, Springer-Verlag Berlin, Berlin, pp. 445–456.
- Gregorich E.G., Rochette P., VandenBygaart A.J., Angers D.A. (2005) Greenhouse gas contributions of agricultural soils and potential mitigation practices in Eastern Canada, Soil Till. Res. 83, 53–72.
- Guarino A., Fabbri C., Brambilla M., Valli L., Navarotto P. (2006) Evaluation of simplified covering systems to reduce gaseous emissions from livestock manure storage, Trans. ASABE 49, 737–747.

- Hansen S. (1996) Effects of manure treatment and soil compaction on plant production of a dairy farm system converting to organic farming practice, Agr. Ecosyst. Environ. 56, 173–186.
- Hansen M.N., Sommer S.G., Henriksen K. (2002) Methane emissions from livestock manure - effects of storage conditions and climate, in: Petersen S., Olesen J. (Eds.), GHG inventories for agriculture in the nordic countries. DIAS (Danish Institute of Agricultural Science), Helsingor, Denmark, pp. 45–53.
- Häring A.M. (2003) Organic dairy farms in the EU: production systems, economics and future development, Livest. Prod. Sci. 80, 89–97.
- Hays M.D., Fine P.M., Geron C.D., Kleeman M.J., Gullett B.K. (2005) Open burning of agricultural biomass: Physical and chemical properties of particle-phase emissions, Atmos. Environ. 39, 6747–6764.
- Hensen A., Groot T.T., van den Bulk W.C.M., Vermeulen A.T., Olesen J.E., Schelde K. (2006) Dairy farm CH<sub>4</sub> and N<sub>2</sub>O emissions, from one square metre to the full farm scale, Agr. Ecosyst. Environ. 112, 146–152.
- Hindrichsen I.K., Wettstein H.R., Machmuller A., Kreuzer M. (2006) Methane emission, nutrient degradation and nitrogen turnover in dairy cows and their slurry at different milk production scenarios with and without concentrate supplementation, Agr. Ecosyst. Environ. 113, 150–161.
- Holland J.M. (2004) The environmental consequences of adopting conservation tillage in Europe: reviewing the evidence, Agr. Ecosyst. Environ. 103, 1–25.
- Hu S., Grunwald N.J., van Bruggen A.H.C., Gamble G.R., Drinkwater L.E., Shennan C., Demment M.W. (1997) Shortterm effects of cover crop incroporation on soil carbon pools and nitrogen availability, Soil Sci. Soc. Am. J. 61, 901–911.
- Husted S. (1994) Seasonal variation in methane emission from stored slurry and solid manures, J. Environ. Qual. 23, 585–592.
- INRA (2002) Stocker du carbone dans les sols agricoles en France? Expertise scientifique collective, in: Arrouays D., Balesdent J., Germon J.C., Jayet P.A., Soussana J.F., Stengel P. (Eds.), Institut National de la Recherche Agronomique (INRA), Paris, p. 332.
- IPCC (2007) Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of working group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Parry M.L., Canziani O.F., Palutikof J.P., Van der Linden P.J., Hanson C.E. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Johnson J.M.-F., Franzluebbers A.J., Weyers S.L., Reicosky D.C. (2007) Agricultural opportunities to mitigate greenhouse gas emissions, Environ. Pollut. 150, 107–124.
- Kebreab E., Clark K., Wagner-Riddle C., France J. (2006) Methane and nitrous oxide emissions from Canadian animal agriculture: A review, Can. J. Anim. Sci. 86, 135–158.
- Keppler F., Hamilton J.T.G., Brass M., Rockmann T. (2006) Methane emissions from terrestrial plants under aerobic conditions, Nature 439, 187–191.
- Kern J.S., Johnson M.G. (1993) Conservation tillage impacts on national soil and atmospheric carbon levels, Soil Sci. Soc. Am. J. 57, 200–210.

- Kirchmann H., Bernal M.P. (1997) Organic waste treatment and C stabilization efficiency, Soil Biol. Biochem. 29, 1747–1753.
- Kuikman P.J., Velthof G.L., Oenema O. (2004) Controlling nitrous oxide emissions from agriculture: experience from The Netherlands, in: Hatch D.J., Chadwick D.R., Jarvis S.C., Roker J.A. (Eds.), Controlling nitrogen flows and losses, 12th Nitrogen Workshop, Wageningen Academic Publishers, University of Exeter, UK.
- Kulling D.R., Menzi H., Sutter F., Lischer P., Kreuzer M. (2003) Ammonia, nitrous oxide and methane emissions from differently stored dairy manure derived from grass- and hay-based rations, Nutr. Cycl. Agroecosyst. 65, 13–22.
- Lal R. (2004a) Agricultural activities and the global carbon cycle, Nutr. Cycl. Agroecosyst. 70, 103–116.
- Lal R. (2004b) Soil carbon sequestration to mitigate climate change, Geoderma 123, 1–22.
- Le Mer J., Roger P. (2001) Production, oxidation, emission and consumption of methane by soils: A review, Eur. J. Soil Biol. 37, 25–50.
- Ledgard S.F. (2001) Nitrogen cycling in low input legume-based agriculture, with emphasis on legume/grass pastures, Plant Soil 228, 43–59.
- Ledgard S., Steele K.W. (1992) Biological nitrogen fixation in mixed legume/grass pastures, Plant Soil 141, 137–153.
- Lovett D.K., Shalloo L., Dillon P., O'Mara F.P. (2006) A systems approach to quantify greenhouse gas fluxes from pastoral dairy production as affected by management regime, Agr. Syst. 88, 156–179.
- Malgeryd J. (1998) Technical measures to reduce ammonia losses after spreading of animal manure, Nutr. Cycl. Agroecosyst. 51, 51–57.
- Marshall E.J.P., Moonen A.C. (2002) Field margins in northern Europe: their functions and interactions with agriculture, Agr. Ecosyst. Environ. 89, 5–21.
- Martin C., Morgavi D., Doreau M., Jouany J.P. (2006) Comment réduire la production de méthane chez les ruminants? Fourrages 187, 283–300.
- Martin C., Rouel J., Jouany J.P., Doreau M., Chilliard Y. (2008) Methane output and diet digestibility in response to feeding dairy cows crude linseed, extruded linseed, or linseed oil, J. Anim. Sci. 2642–2650.
- Martinez J., Guiziou F., Peu P., Gueutier V. (2003) Influence of treatment techniques for pig slurry on methane emissions during subsequent storage, Biosyst. Eng. 85, 347–354.
- Mathieu O., Leveque J., Henault C., Milloux M.J., Bizouard F., Andreux F. (2006) Emissions and spatial variability of N<sub>2</sub>O, N<sub>2</sub>and nitrous oxide mole fraction at the field scale, revealed with N-15 isotopic techniques, Soil Biol. Biochem. 38, 941–951.
- McAllister T.A., Okine E.K., Mathison G.W., Cheng K.J. (1996) Dietary, environmental and microbiological aspects of methane production in ruminants, Can. J. Anim. Sci. 76, 231–243.
- McCaughey W.P., Wittenberg K., Corrigan D. (1997) Methane production by steers on pasture, Can. J. Anim. Sci. 77, 519–524.
- McCaughey W.P., Wittenberg K., Corrigan D. (1999) Impact of pasture type on methane production by lactating beef cows, Can. J. Anim. Sci. 79, 221–226.

- McGinn S.M. (2006) Measuring greenhouse gas emissions from point sources in agriculture, Can. J. Soil Sci. 86, 355–371.
- McNeill A.M., Eriksen J., Bergstrom L., Smith K.A., Marstorp H., Kirchmann H., Nilsson I. (2005) Nitrogen and sulphur management: challenges for organic sources in temperate agricultural systems, Soil Use Manage. 21, 82–93.
- Mills J.A.N., Dijkstra J., Bannink A., Cammell S.B., Kebreab E., France J. (2001) A mechanistic model of whole-tract digestion and methanogenesis in the lactating dairy cow: Model development, evaluation, and application, J. Anim. Sci. 79, 1584–1597.
- Misselbrook T.H., Brookman S.K.E., Smith K.A., Cumby T., Williams A.G., McCrory D.F. (2005) Crusting of stored dairy slurry to abate ammonia emissions: Pilot-scale studies, J. Environ. Qual. 34, 411–419.
- Monteny G.J., Bannink A., Chadwick D. (2006) Greenhouse gas abatement strategies for animal husbandry, Agr. Ecosyst. Environ. 112, 163–170.
- Monteny G.J., Groenestein C.M., Hilhorst M.A. (2001) Interactions and coupling between emissions of methane and nitrous oxide from animal husbandry, Nutr. Cycl. Agroecosyst. 60, 123–132.
- Morvan T., Nicolardot B., Pean L. (2006) Biochemical composition and kinetics of C and N mineralization of animal wastes: a typological approach, Biol. Fert. Soils 42, 513–522.
- Mosier A., Kroeze C., Nevison C., Oenema O., Seitzinger S., van Cleemput O. (1998) Closing the global N<sub>2</sub>O budget: nitrous oxide emissions through the agricultural nitrogen cycle, Nutr. Cycl. Agroecosyst. 52, 225–248.
- Mosimann E., Suter D. (2003) Autonomie en protéines et environnement: le compromis helvétique, Fourrages 175, 333–345.
- Mustin M. (1987) Le compost Gestion de la matière organique, Dubusc F., 954 p.
- Nicholas P.K., Padel S., Cuttle S.P., Fowler S.M., Hovi M., Lampkin N.H., Weller R.F. (2004) Organic dairy production: a review, Biol. Agric. Hortic. 22, 1–58.
- Oenema O., Velthof G., Kuikman P. (2001) Technical and policy aspects of strategies to decrease greenhouse gas emissions from agriculture, Nutr. Cycl. Agroecosyst. 60, 301–315.
- Oenema O., Velthof G.L., Yamulki S., Jarvis S.C. (1997) Nitrous oxide emissions from grazed grassland, Soil Use Manage. 13, 288–295.
- Oenema O., Wrage N., Velthof G.L., van Groenigen J.W., Dolfing J., Kuikman P.J. (2005) Trends in global nitrous oxide emissions from animal production systems, Nutr. Cycl. Agroecosyst. 72, 51–65.
- Olesen J.E., Schelde K., Weiske A., Weisbjerg M.R., Asman W.A.H., Djurhuus J. (2006) Modelling greenhouse gas emissions from European conventional and organic dairy farms, Agr. Ecosyst. Environ. 112, 207–220.
- Peigne J., Girardin P. (2004) Environmental impacts of farmscale composting practices, Water Air Soil Pollut. 153, 45–68.
- Petersen S.O. (1999) Nitrous oxide emissions from manure and inorganic fertilizers applied to spring barley, J. Environ. Qual. 28, 1610–1618.
- Petersen S.O., Amon B., Gattinger A. (2005) Methane oxidation in slurry storage surface crusts, J. Environ. Qual. 34, 455–461.

- Petersen S.O., Regina K., Pollinger A., Rigler E., Valli L., Yamulki S., Esala M., Fabbri C., Syvasalo E., Vinther F.P. (2006) Nitrous oxide emissions from organic and conventional crop rotations in five European countries, Agr. Ecosyst. Environ. 112, 200–206.
- Pflimlin A., Kempf M. (2002) Trends of the organic dairy industry in some European countries, 9<sup>es</sup> Rencontres autour des Recherches sur les Ruminants, Paris, France, 4–5 decembre 2002, pp. 215–218.
- Pinares-Patino C.S., Baumont R., Martin C. (2003) Methane emissions by Charolais cows grazing a monospecific pasture of timothy at four stages of maturity, Can. J. Anim. Sci. 83, 769–777.
- Pinares-Patino C.S., D'Hour P., Jouany J.-P., Martin C. (2007) Effects of stocking rate on methane and carbon dioxide emissions from grazing cattle, Agr. Ecosyst. Environ. 121, 30–46.
- Rasmussen P.E., Douglas C.L., Collins H.P., Albrecht S.L. (1998) Long-term cropping system on mineralizable nitrogen in soil, Soil Biol. Biochem. 30, 1829–1837.
- Rees R.M., Bingham I.J., Baddeley J.A., Watson C.A. (2005) The role of plants and land management in sequestering soil carbon in temperate arable and grassland ecosystems, Geoderma 128, 130–154.
- Reicosky D.C., Reeves D.W., Prior S.A., Runion G.B., Rogers H.H., Raper R.L. (1999) Effects of residue management and controlled traffic on carbon dioxide and water loss, Soil Till. Res. 52, 153–165.
- Rochette P., Janzen H.H. (2005) Towards a revised coefficient for estimating N<sub>2</sub>O emissions from legumes, Nutr. Cycl. Agroecosyst. 73, 171–179.
- Rubaek G.H., Henriksen K., Petersen J., Rasmussen B., Sommer S.G. (1996) Effects of application technique and anaerobic digestion on gaseous nitrogen loss from animal slurry applied to ryegrass (Lolium perenne), J. Agr. Sci. 126, 481–492.
- Saggar S., Bolan N.S., Bhandral R., Hedley C.B., Luo J. (2004) A review of emissions of methane, ammonia, and nitrous oxide from animal excreta deposition and farm effluent application in grazed pastures, New Zeal J. Agr. Res. 47, 513–544.
- Schils R.L.M., Olesen J.E., del Prado A., Soussana J.F. (2007) A review of farm level modelling approaches for mitigating greenhouse gas emissions from ruminant livestock systems, Livest. Sci. 112, 240–251.
- Schils R.L.M., Verhagen A., Aarts H.F.M., Kuikman P.J., Sebek L.B.J. (2006) Effect of improved nitrogen management on greenhouse gas emissions from intensive dairy systems in the Netherlands, Global Change Biol. 12, 382–391.
- Schils R.L.M., Verhagen A., Aarts H.F.M., Sebek L.B.J. (2005) A farm level approach to define successful mitigation strategies for GHG emissions from ruminant livestock systems, Nutr. Cycl. Agroecosyst. 71, 163–175.
- Shepherd M.A. (2000) The environmental implications of manure use in organic farming systems, ADAS, p. 20.
- Shepherd M., Pearce B., Philipps L., Cuttle S., Bhogal A., Costigan P., Unwin R. (2003) An assessment of the environmental impacts of organic farming, DEFRA, p. 80.
- Six J., Ogle S.M., Breidt F.J., Conant R.T., Mosier A.R., Paustian K. (2004) The potential to mitigate global warming with notillage management is only realized when practised in the long term, Global Change Biol. 10, 155–160.

- Smith P. (2004) Carbon sequestration in croplands: the potential in Europe and the global context, Eur. J. Agron. 20, 229–236.
- Smith K.A., Conen F. (2004) Impacts of land management on fluxes of trace greenhouse gases, Soil Use Manage. 20, 255–263.
- Smith P., Goulding K.W., Smith K.A., Powlson D.S., Smith J.U., Falloon P., Coleman K. (2001) Enhancing the carbon sink in European agricultural soils: including trace gas fluxes in estimates of carbon mitigation potential, Nutr. Cycl. Agroecosyst. 60, 237–252.
- Smith P., Martino D., Cai Z.C., Gwary D., Janzen H., Kumar P., McCarl B., Ogle S., O'Mara F., Rice C., Scholes B., Sirotenko O., Howden M., McAllister T., Pan G.X., Romanenkov V., Schneider U., Towprayoon S. (2007) Policy and technological constraints to implementation of greenhouse gas mitigation options in agriculture, Agr. Ecosyst. Environ. 118, 6–28.
- Smith P., Martino D., Cai Z., Gwary D., Janzen H., Kumar P., McCarl B., Ogle S., O'Mara F., Rice C., Scholes B., Sirotenko O., Howden M., McAllister T., Pan G.X., Romanenkov V., Schneider U., Towprayoon S., Wattenbach M., Smith J. (2008) Greenhouse gas mitigation in agriculture, Philos. T. Roy. Soc. B-Biol. Sci. 363, 789–813.
- Smith P., Milne R., Powlson D.S., Smith J.U., Falloon P., Coleman K. (2000) Revised estimates of the carbon mitigation potential of UK agricultural land, Soil Use Manage. 16, 293–295.
- Smith P., Powlson D.S., Glendining M.J., Smith J.U. (1997) Potential for carbon sequestration in European soils: preliminary estimates for five scenarios using results from long-term experiments, Global Change Biol. 3, 67–79.
- Sneath R.W., Beline F., Hilhorst M.A., Peu P. (2006) Monitoring GHG from manure stores on organic and conventional dairy farms, Agr. Ecosyst. Environ. 112, 122–128.
- Sommer S.G. (2001) Effect of composting on nutrient loss and nitrogen availability of cattle deep litter, Eur. J. Agron. 14, 123–133.
- Sommer S.G., Hutchings N.J. (2001) Ammonia emission from field applied manure and its reduction - invited paper, Eur. J. Agron. 15, 1–15.
- Sommer S.G., Olesen J.E. (2000) Modelling ammonia emission from animal slurry applied with trail hoses to cereals, Atmos. Environ. 34, 2361–2372.
- Sommer S.G., Christensen B.T., Nielsen N.E., Schjorring J.K. (1993) Ammonia volatilization during storage of cattle and pig slurry: effect of surface cover, J. Agr. Sci. 121, 63–71.
- Sommer S.G., Petersen S.O., Sogaard H.T. (2000) Greenhouse gas emission from stored livestock slurry, J. Environ. Qual. 29, 744–751.
- Sorensen P., Weisbjerg M.R., Lund P. (2003) Dietary effects on the composition and plant utilization of nitrogen in dairy cattle manure, J. Agr. Sci. 141, 79–91.
- Soussana J.F., Allard V., Pilegaard K., Ambus P., Amman C., Campbell C., Ceschia E., Clifton-Brown J., Czobel S., Domingues R., Flechard C., Fuhrer J., Hensen A., Horvath L., Jones M., Kasper G., Martin C., Nagy Z., Neftel A., Raschi A., Baronti S., Rees R.M., Skiba U., Stefani P., Manca G., Sutton M., Tuba Z., Valentini R. (2007) Full accounting of the greenhouse gas (CO<sub>2</sub>, N<sub>2</sub>O, CH<sub>4</sub>) budget of nine European grassland sites, Agr. Ecosyst. Environ. 121, 121–134.

- Soussana J.F., Loiseau P., Vuichard N., Ceschia E., Balesdent J., Chevallier T., Arrouays D. (2004) Carbon cycling and sequestration opportunities in temperate grasslands, Soil Use Manage. 20, 219–230.
- Steinfeld H., Gerber P., Wassenaar T., Castel V., Rosales M., de Haan C. (2006) Livestock's long shadow - environmental issues and options. in: FAO (Ed.), FAO, Rome, p. 390.
- Steinshamn H., Thuen E., Bleken M.A., Brenoe U.T., Ekerholt G., Yri C. (2004) Utilization of nitrogen (N) and phosphorus (P) in an organic dairy farming system in Norway, Agr. Ecosyst. Environ.104, 509–522.
- Stockdale E.A., Rees R.M. (1995) Release of nitrogen from plant and animal residues and consequent plant uptake efficiency, Biol. Agric. Hortic. 11, 229–245.
- Stockdale E.A., Lampkin N.H., Hovi M., Keatinge R., Lennartsson E.K.M., Macdonald D.W., Padel S., Tattersall F.H., Wolfe M.S., Watson C.A. (2001) Agronomic and environmental implications of organic farming systems, Adv. Agron. 70, 261–327.
- Stolze M., Piorr A., Haring A., Dabbert S. (2000) The Environmental Impacts of Organic Farming in Europe, in: Dabbert S., Lampkin N., Michelsen J., Nieberg H., Zanoli R. (Eds.), Organic Farming in Europe: Economics and Policy. Dabbert S. - University of Hohenheim, Stuttgart-Hohenheim, p. 127.
- Stopes C., Lord E.I., Philipps L., Woodward L. (2002) Nitrate leaching from organic farms and conventional farms following best practice, Soil Use Manage. 18, 256–263.
- Sutton M.A., Milford C., Nemitz E., Theobald M.R., Hill P.W., Fowler D., Schjoerring J.K., Mattsson M.E., Nielsen K.H., Husted S., Erisman J.W., Otjes R., Hensen A., Mosquera J., Cellier P., Loubet B., David M., Genermont S., Neftel A., Blatter A., Herrmann B., Jones S.K., Horvath L., Fuhrer E.C., Mantzanas K., Koukoura Z., Gallagher M., Williams P., Flynn M., Riedo M. (2001) Biosphere-atmosphere interactions of ammonia with grasslands: Experimental strategy and results from a new European initiative, Plant Soil 228, 131–145.
- Sutton M.A., Pitcairn C.E.R., Fowler D. (1993) The exchange of ammonia between the atmosphere and plant communities, Adv. Ecol. Res. 24, 301–393.
- Syväsalo E., Regina K., Turtola E., Lemola R., Esala M. (2006) Fluxes of nitrous oxide and methane, and nitrogen leaching from organically and conventionally cultivated sandy soil in western Finland, Agr. Ecosyst. Environ. 113, 342–348.
- Tavendale M.H., Lane G.A., Schreurs N.M., Fraser K., Meagher L.P. (2005) The effects of condensed tannins from Dorycnium rectum on skatole and indole ruminal biogenesis for grazing sheep, Aust. J. Agr. Res. 56, 1331–1337.
- Theobald M.R., Milford C., Hargreaves K.J., Sheppard L.J., Nemitz E., Tang Y.S., Phillips V.R., Sneath R., McCartney L., Harvey F.J., Leith I.D., Cape J.N., Fowler D., Sutton M.A. (2002) Potential for ammonia recapture by farm woodlands: design and application of a new experimental facility, in: Optimizing nitrogen management in food and energy production and environmental protection, 2nd International Nitrogen Conference, Potomac, Maryland, USA, 14–18 October 2001, A.A. Balkema Publishers, Lisse Netherlands.
- Vaarst M., Padel S., Hovi M., Younie D., Sundrum A. (2005) Sustaining animal health and food safety in European organic livestock farming, Livest. Prod. Sci. 94, 61–69.

- Vellinga T.V., van den Pol-van Dasselaar A., Kuikman P.J. (2004) The impact of grassland ploughing on CO<sub>2</sub> and N<sub>2</sub>O emissions in the Netherlands, Nutr. Cycl. Agroecosyst. 70, 33–45.
- Velthof G.L., van Beusichem M.L., Oenema O. (1998) Mitigation of nitrous oxide emission from dairy farming systems, Environ. Pollut. 102, 173–178.
- Velthof G.L., Kuikman P.J., Oenema O. (2002) Nitrous oxide emission from soils amended with crop residues, Nutr. Cycl. Agroecosyst. 62, 249–261.
- Vleeshouwers L.M., Verhagen A. (2002) Carbon emission and sequestration by agricultural land use: a model study for Europe, Global Change Biol. 8, 519–530.
- Watson C.A., Atkinson D., Gosling P., Jackson L.R., Rayns F.W. (2002) Managing soil fertility in organic farming systems, Soil Use Manage. 18, 239–247.
- Watson C.A., Oborn I., Eriksen J., Edwards A.C. (2005) Perspectives on nutrient management in mixed farming systems, Soil Use Manage. 21, 132–140.
- Webb J., Chadwick D., Ellis S. (2004) Emissions of ammonia and nitrous oxide following incorporation into the soil of farmyard manures stored at different densities, Nutr. Cycl. Agroecosyst. 70, 67–76.
- Webb J., Menzi H., Pain B.F., Misselbrook T.H., Dammgen U., Hendriks H., Dohler H. (2005) Managing ammonia emissions from livestock production in Europe, Environ. Pollut. 135, 399–406.
- Weiske A., Vabitsch A., Olesen J.E., Schelde K., Michel J., Friedrich R., Kaltschmitt M. (2006) Mitigation of greenhouse gas emissions in European conventional and organic dairy farming, Agr. Ecosyst. Environ. 112, 221–232.

- Weller R.F. (2002) A comparison of two systems of organic milk production, in: Kyriazakis Z. (Ed.), Proceedings of Organic Meat and Milk for Ruminants, EAAP Publication, Athens, pp. 111–116.
- West T.O., Post W.M. (2002) Soil organic carbon sequestration rates by tillage and crop rotation: A global data analysis, Soil Sci. Soc. Am. J. 66, 1930–1946.
- West T.O., Six J. (2007) Considering the influence of sequestration duration and carbon saturation on estimates of soil carbon capacity, Climatic Change 80, 25–41.
- WHO (2004) Health aspects of air pollution: Results from the WHO project systematic review of health aspects of air pollution in Europe, p. 24.
- Woodward S.L., Waghorn G.C., Lassey K.R., Laboyrie P.G. (2002) Does feeding sulla (Hedysarum coronarium) reduce methane emissions from dairy cows? Proc. N.Z. Soc. Anim. Prod. 62, 227–230.
- Wright A.L., Dou F., Hons F.M. (2007) Crop species and tillage effects on carbon sequestration in subsurface soil, Soil Sci. 172, 124–131.
- Wulf S., Maeting M., Clemens J. (2002a) Application technique and slurry co-fermentation effects on ammonia, nitrous oxide, and methane emissions after spreading: II. Greenhouse gas emissions, J. Environ. Qual. 31, 1795–1801.
- Wulf S., Maeting M., Clemens J. (2002b) Application technique and slurry co-fermentation effects on ammonia, nitrous oxide, and methane emissions after spreading: I. Ammonia volatilization, J. Environ. Qual. 31, 1789–1794.
- Yamulki S. (2006) Effect of straw addition on nitrous oxide and methane emissions from stored farmyard manures, Agr. Ecosyst. Environ. 112, 140–145.

# Water Deficit and Nitrogen Nutrition of Crops

Victoria Gonzalez-Dugo, Jean-Louis Durand, and François Gastal

### Contents

1	Introd	luction	558		
2	Quant	tification of Plant Water and Nitrogen	560		
	<b>Nutri</b> 2.1	Plant Water Status	560		
	2.1	Crop and Plant Nitrogen Status	560		
3	Effect	of Water Deficit on Nitrogen Nutrition			
	Dynai	mic	561		
4	Effect	of Water Deficit on Plant Nitrogen Demand .	563		
	4.1	Growth of the Whole-Plant Biomass	563		
	4.2	Root Shoot Ratio and Nitrogen Demand	563		
5	Effect	of Water Deficit on Nitrogen Supply	564		
	5.1	Effect on Soil N Transformation Processes	564		
	5.2	Soil Fluxes: Transport of Nitrogen to the			
		Roots, Mass Flow and Diffusion	564		
	5.3	Plant Physiological Processes Involved in the			
		Definition of Nitrogen Supply	565		
6	Respo	nse of Crop Physiological Functions			
	Involv	red in Nitrogen Nutrition to Water Deficit	566		
	6.1	Nitrogen Assimilation in Plant Tissues	566		
	6.2	Effect on Nitrogen Movement Through the			
		Xylem	567		
	6.3	Remobilization of Nitrogen to Grains	567		
7	Conce	epts Required for Simulation Models	568		
8	Concl	usion	569		
<b>References</b>					

V. Gonzalez-Dugo (🖂)

INRA, Unité de Recherche Pluridisciplinaire sur les prairies et les plantes fourragères, BP 6, 86600 Lusignan, France

and

IAS-CSIC, Instituto de Agricultura Sostenible, Consejo Superior de Investigaciones Científicas, Alameda del Obispo, s/n, 14004 Córdoba, Spain e-mail: victoria.gonzalez@ias.csic.es Abstract Among the environmental factors that can be modified by farmers, water and nitrogen are the main ones controlling plant growth. Irrigation and fertilizer application overcome this effect, if adequately used. Agriculture thus consumes about 85% of the total fresh water used worldwide. While only 18% of the world's cultivated areas are devoted to irrigated agriculture, this total surface represents more than 45% of total agricultural production. These data highlight the importance of irrigated agriculture in a framework where the growing population demands greater food production. In addition, tighter water restrictions and competition with other sectors of society is increasing pressure to diminish the share of fresh water for irrigation, thus resulting in the decrease in water diverted for agriculture. The effect of water and nutrient application on yield has led to the overuse of these practices in the last decades. This misuse of irrigation and fertilizers is no longer sustainable, given the economic and environmental costs. Sustainable agriculture requires a correct balance between the agronomic, economic and environmental aspects of nutrient management. The major advances shown in this review are the following: (1) the measurement of the intensity of drought and N deficiency is a prerequisite for quantitative assessment of crop needs and management of both irrigation and fertilizer application. The N concentration of leaves exposed to direct irradiance allows both a reliable and high-resolution measurement of the status and the assessment of N nutrition at the plant level. (2) Two experiments on sunflower and on tall fescue are used to relate the changes in time and irrigation intensity to the crop N status, and to introduce the complex relationships between N demand and supply in crops. (3) Effects of water deficits on N demand are reviewed,

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_25, © Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences

from Gonzalez-Dugo et al., Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2009059

pointing out the high sensitivity of N-rich organs versus the relative lesser sensitivity of organs that are poorer in N compounds. (4) The generally equal sensitivities of nitrifying and denitrifying microbes are likely to explain many conflicting results on the impact of water deficits on soil mineral N availability for crops. (5) The transpiration stream largely determines the availability of mineral N in the rhizosphere. This makes our poor estimate of root densities a major obstacle to any precise assessment of N availability in fertilized crops. (6) The mineral N fluxes in the xylem are generally reduced under water deficit and assimilation is generally known to be more sensitive to water scarcity. (7) High osmotic pressures are maintained during grain filling, which enables the plant to recycle large amounts of previously assimilated N. Its part in the total grain N yield is therefore generally higher under water deficits. (8) Most crop models currently used in agronomy use N and water efficiently but exhibit different views on their interaction.

**Keywords** Drought • Nitrogen nutrition status • Supply • Demand • Balance

### **1** Introduction

Water and N are the most limiting factors in agricultural production in most parts of the world, especially in arid and semiarid zones. Given current and future needs for high yields (Cassman, 2001; Sheehy, 2001), concerns are now being raised regarding the soil resources necessary to meet human needs. Traditionally, water and N have been overused in agriculture for decades, but this is no longer sustainable, considering the economic and environmental costs of these practices (Addiscott et al., 1991; Jury and Vaux, 2005). More than 55% of the increase in crop production, especially in emerging countries, comes from the use of chemical fertilizers, with N fertilizers being dominant (Li et al., 2009). However, N is also a significant pollutant that has a great impact on ecosystem deterioration (Galloway et al., 2008) and biodiversity (Stevens et al., 2004). Water scarcity, pollution and energy consumption are driving the maximization of water and N use in agriculture, in order to meet current and future demand while reducing resource requirements.

Climate change is an additional factor that increases uncertainty over the supply of resources (Asseng et al., 2009). Rainfall patterns are expected to change across the world (Giorgi and Bi, 2005), and for some regions, including most of Europe, droughts will be more frequent and severe (Easterling et al., 2000). This means that irrigation water will be more necessary than before to maintain production levels, and that crops will face water scarcity more often.

In situations where water availability cannot be assured, a plant can be subjected to a water deficit that may diminish growth. There are many physiological processes that are affected by water availability, mineral nutrition among others (Lemaire and Denoix, 1987; Onillon et al., 1995; Gonzalez-Dugo et al., 2005). As the major transport agent, and given its central role in biology, water determines the entire N biogeochemical cycle and, ultimately, its availability for plant production. The purpose of this work is to review the major advances in the assessment of the effect of water deficit on N nutrition.

The effect of water deficit on N nutrition has been the subject of considerable research at both plant (Morgan, 1984; Nicolas et al., 1985; Larsson, 1992; Matzner and Richards, 1996) and canopy (among others, Lemaire and Denoix, 1987; Cantero-Martinez et al., 1995; Onillon et al., 1995; Gonzalez-Dugo et al., 2005, Mistele and Schmidhalter, 2008) levels, but its involvement in a large number of interconnected metabolic processes makes it difficult to predict the effect of water deficit on plant N nutrition status. It is therefore necessary to consider the series of biophysical and chemical processes to which N is subjected, in different chemical forms and through transfers between living organisms and abiotic components in an ecosystem.

 $N_2$  fixation in leguminous plants is also known to be affected by water scarcity. Symbiotic fixation will not be considered within this work. For more information, see Streeter (2003) and Thomas et al. (2004).

During its life cycle, a plant may be subjected to a water deficit, a N deficit or a combination of both, thus co-limiting (Sadras, 2005) its productivity. The effects of water and/or N deficit on plant production therefore depend on their timing and intensity (Bradford and Hsiao, 1982; Nielsen and Halvorson, 1991). Irrigation and fertilizer application increase yield when soil water and N levels are limited. Due to increasing water shortages at the global level and groundwater



Fig. 1 Conceptual framework of the principal processes involved in the response of crop N nutrition to water deficit. Plain arrows indicate fluxes (mass or energy), while dotted arrows indicate direct causal relationships

pollution, while demand for food grows, it is essential to maximize the yield per unit water and N applied.

Crop N uptake has often been considered in relation either to soil availability (soil N supply approach) or to crop growth (N demand approach). It actually results from both (Gastal and Lemaire, 2002). Nitrogen absorption by crops is automatically reduced under dry conditions, even when mineral N is present in the soil colonized by roots (Gonzalez-Dugo et al., 2005). The absorption of N by roots requires the presence of water in the soil, as it is the agent that transports solutes to the soil-root interface (Garwood and Williams, 1967). Under water scarcity, N demand by plants is reduced, as growth rate is diminished. If the effect on N supply is greater than that on plant growth, the result will be a N deficiency. The implications of this demand/supply duality can be observed when different species exhibiting contrasted growth rates are compared under similar conditions (Gonzalez-Dugo et al., 2005). This deterioration of N nutrition also implies a reduction in growth in addition to water deficit-induced reduction. This series of direct and indirect (via N limitation) effects of water shortage on plant growth makes any quantification of the effect of water deficit on N nutrition a complex task. The correct assessment of nutrition status by indices and the use of physiology-based simulation models are therefore valuable tools for the quantitative assessment of direct and N-induced water effects on growth.

The influence of water deficit on N nutrition status is at play at the level of the soil (through the availability of mineral N for root uptake) as well as at plant level. However, it is difficult to ascribe a ranking to any of these levels in terms of their effects because water availability for transpiration, carbon supply and growth potential all determine N demand, assimilation and distribution within the plant. These complex relationships are summarized in Fig. 1, where the most important processes located in the soil and plant are shown. This review focuses on the principal processes affected. It first describes the main features of an operational definition of plant N nutrition status, in order to assess plant N demand correctly. It then analyzes the N biogeochemical cycle from the soil to its incorporation in the plant structure. Finally, some of the most used physiology-based simulation models are reviewed in order to analyze how the effect of water and N interaction on plant growth is modeled.

# 2 Quantification of Plant Water and Nitrogen Nutrition Status

# 2.1 Plant Water Status

The measurement of plant water status in terms of the water potential of leaves ( $\psi$ ; MPa) directly generates a variable that is relevant to both water absorption and plant functioning, related specifically to the turgor pressure of cells (P; MPa):

$$\psi = \mathbf{P} - \pi \tag{1}$$

where  $\pi$  is the osmotic pressure of leaf tissues (MPa). The absorption of water into the plant, and its transpiration, are directly proportional to the fall in  $\psi$  values throughout the soil plant atmosphere continuum. This approach (Van Der Honert, 1948) has been widely validated and has proved its ability to explain most changes in plant water status in the field, because the main factors involved in water movements are fundamentally physical (Dixon and Joly, 1895; Phillip, 1966; Wei et al., 1999; Tyree and Cochard, 2003). Furthermore, physiological variables relevant to the plant ( $\pi$  and P) are directly related to the ratio between the actual water volume of the plant and its maximum volume (at full turgor and  $\psi = 0$ ). The plant water status is hence best measured in most cases by the mean leaf water potential.

One of the main limitations for the assessment of water status using leaf water potential is that it is very time-consuming. Remote sensing-derived techniques are opening new avenues in assessing plant nutritional status (Berni et al., 2009). Thermal and multispectral information acquired by sensors onboard satellite or airborne platforms provides a sound basis for developing indices to assess plant water status, such as the Crop Water Stress Index (Idso et al., 1981; Jackson et al., 1981), which is closely related to the canopy temperature and stomatal conductance (Berni et al., 2009). Other indices have been developed from spectral information. The Photochemical Reflectance Index (Gamon et al., 1992), which is related to the xanthophyll cycle, has demonstrated a strong relationship with water status and canopy temperature (Suarez et al., 2009). A major reason for the interest in these techniques is the possibility of remote measurement, which helps avoid time-consuming techniques, such as water potential or stomatal conductance, used for assessing water status at the field or

### 2.2 Crop and Plant Nitrogen Status

farm level.

The assessment of plant N nutrition status is far more complex than that of plant water status, because both in-soil and in-plant biological processes directly interfere with N availability and uptake, and even under optimum conditions, N concentration in the plant diminishes continuously during crop development (Van Dobben, 1962; Angus and Mancur, 1985; Van Keulen and Seligman, 1987). The cause of this decline is linked firstly to N dilution within the plant volume. Plant N is mainly located in metabolically active aerial plant parts, and is thus related more directly to plant surface than to plant volume. As a plant grows, the proportion (in weight) of structural and storage tissues, which are relatively poor in N, increases. As a consequence, the average plant N concentration diminishes (Greenwood et al., 1990; Lemaire and Gastal, 1997). At the crop level, the relationship between plant N content and its surface also results from the close relationship between the shoot N concentration and the incident solar irradiance (Hardwick, 1987; Gastal and Lemaire, 2002). Within the canopy, leaves exposed directly to solar radiation have the highest N concentration, which optimizes N use relative to carbon assimilation (Hirose and Werger, 1987; Hikosaka et al., 1994). Three-quarters of total reduced N in the leaf may be connected with photosynthesis (Field and Mooney, 1986), mostly in the form of RuBP carboxylase and chlorophyll. In order to measure plant N nutrition status, several indices have been designed in recent decades. Overman et al... (1995) developed a model where growth was related to fertilizer application. However, the actual amount of N available for the plant, including mineralized soil organic N, is not considered, making that approach difficult to extrapolate to different scenarios (Jeuffroy et al., 2002). In spite of their wide range of possible applications, indices considering soil N content only do not seem suited for assessing plant N status. Nitrogen concentration or uptake could not represent by itself the plant's N nutrition status, as it is directly related to plant biomass and growth rate (Lemaire

and Gastal, 2009). In a controlled environment, it was shown that the relationship between N uptake ( $NO_3^$ or  $NH_4^+$ ) and the ionic concentration in soil solution followed a hyperbolic-type law with several kinetic phases (Rao and Rains, 1976; Macduff et al., 1989; Tischner, 2000; Glass et al., 2002). Devienne-Barret et al. (2000) showed that such relationships also held true in the field. Meanwhile, Lemaire and Meynard (1997) pointed out that the N available for plant uptake is not only defined by the amount of mineral N contained in the rooting zone, but also by soil conditions defining root growth and activity.

At the crop level, Lemaire and Salette (1984) studied the relationship between shoot N content (in %) and dry matter content in tall fescue and cocksfoot plots cut to a stubble height of 5 cm. They defined a "critical N concentration" as the minimal N concentration required to achieve a maximal growth rate. Using the data obtained at several N fertilizer application rates, they showed that the critical N concentration was a function of aboveground biomass, which was statistically fitted to an allometric function:

$$N_c = a \cdot W^{-b} \tag{2}$$

where  $N_c$  is the critical N concentration (%) in the dry matter yield (W, in T ha<sup>-1</sup>), and a and b are two parameters (equal to 4.8 and 0.36, respectively) for a tall fescue crop. Similar values were found in a number of other crop species (e.g. wheat: Justes et al., 1994; Mistele and Schmidhalter, 2008; pea: Ney et al., 1997; rape: Colnenne et al., 1998; potato: Bélanger et al., 2001; maize and sorghum: Greenwood et al., 1990; maize: Plénet and Lemaire, 1999). In all these species, the main difference was clearly related to the two metabolic pathways, C3 and C4 (Greenwood et al., 1990). According to this, the Nitrogen Nutrition Index proposed (I<sub>NN</sub>, Eq. (3)) is the ratio between the actual shoot N concentration (N<sub>o</sub>) and the critical value given for its shoot dry weight (Lemaire and Gastal, 1997).

$$I_{\rm NN} = \frac{N_{\rm o}}{N_{\rm c}} \tag{3}$$

where N is non-limiting for growth when the Nitrogen Nutrition Index is larger than 1. The N Nutrition Index is subject to considerable inertia as it requires a significant quantity of biomass and it is difficult to keep accurate records over time, as the surface area required to measure yield is relatively large. This is a major drawback when trying to link rapid changes in water status to crop N status. Another limitation when determining the Nitrogen Nutrition Index is that it only applies to crop level and does not provide a direct and accurate estimate of the N status of individual plants.

At the plant level, it will be necessary to distinguish between the N nutrition status of different elements (species, populations or individual plants with different water status) within a complex crop, as in intercropping systems (e.g. in vine-grass systems, Celette et al., 2005) or in forage production. A solution to this problem was recently proposed by Farruggia et al. (2004), who found a linear relationship between the Nitrogen Nutrition Index (I<sub>NN</sub>) and the N concentration of leaves directly exposed to incident radiation (N<sub>sup</sub>), i.e., lamina at the top of the canopy:

$$N_{sup} = 3 \times I_{NN} + 0.97.$$
 (4)

Duru (2004) fully confirmed this relationship in cocksfoot. With this method, the Nitrogen Nutrition Index can be determined rapidly, and it allows separate analysis of the N status of species growing together on the same plot, provided they have some leaves in the top layer of the canopy. It also increases the possibility of assessing the N status of a crop in situations where its growth is very limited, as is often the case with water deficits. Other authors recently found a relationship between the Nitrogen Nutrition Index and the chlorophyll and polyphenol contents in wheat, which can be used as an indicator of the Nitrogen Nutrition Index (Cartelat et al., 2005; Rodriguez et al., 2009). Accurate tools are thus already available for the assessment of water and N nutrition status, which is a sine qua non condition for elucidating their interaction and its final effect on growth and development.

# 3 Effect of Water Deficit on Nitrogen Nutrition Dynamic

There is some evidence in the literature that shows how N status is altered by water deficit, both in the shorter and the longer term. Using the tabulated data generated by Alvarez de Toro (1987) during an irrigation experiment on sunflower, the relationship between shoot

5 **Dilution curve** N concentration in plant shoots (%) 4 3 2 80%ET 30%ET ٢ı 0%ET 0 5 10 0 15 Shoot biomass (t/ha)

**Fig. 2** Relationship between shoot biomass and N concentration during the pre-anthesis period under four water treatments, ranging from 0 to 80% of the fully irrigated level (FI), with a N application rate of 210 kg N·ha<sup>-1</sup>. The dotted line corresponds to the dilution curve of equation (2) for tall fescue and cocksfoot (Lemaire and Salette, 1984). Based on tabulated data from Alvarez de Toro (1987)

growth and N content during the pre-anthesis period was established (Fig. 3). From this work, we have selected three watering treatments (applying 0, 30 and 80% of evapotranspiration requirements, ET) receiving 210 kg N/ha. Despite the lack of any published references concerning the suitability of the critical N curve in sunflower, this figure clearly shows that for the same quantity of N applied to the soil, and with lower shoot biomass, especially on the last measurement date, N nutrition status declined under drier treatments (0% ET and 30% ET). A similar response was found with less fertilizer application, but the differences between irrigation regimes were smaller (not shown).

In tall fescue, Onillon et al. (1995) analyzed the effect of drought on the relationship between shoot growth and shoot N content from which the Nitrogen Nutrition Index could be computed (Fig. 2). Dry treatments generally exhibited lower Nitrogen Nutrition Index values than irrigated treatments. This effect was more pronounced when the amount of available N was higher. The authors found that in most cases, rewatering and the recovery of crop water status increased N status to values that were similar to those measured on



Nitrogen Nutrition Index

High N, irrigated

1.4

1.2

**Fig. 3** Changes over time in the Nitrogen Nutrition Index during summer regrowth of a tall fescue sward. Four treatments were studied: high (circles) and low (squares) N applications. Black and white symbols show the irrigated and dry treatments, respectively (Onillon et al., 1995)

irrigated plots. Gonzalez-Dugo et al. (2005) showed the variability in the resulting N nutrition status (established by means of the Nitrogen Nutrition Index) in two forage species displaying contrasting growth rate and root architecture (and sown on the same soil) when water and N were both limited. Italian ryegrass had a slower growth rate with a high percentage of fine roots in upper layers, while tall fescue showed a higher growth rate, but with thicker and deeper roots. Tall fescue was found to be more sensitive to water shortageinduced N deficiency, as its growth rate was higher and its root system performed less well in N acquisition. The importance of biomass and plant N remobilization in the inertia of N nutrition status as well as of the root system distribution was highlighted. The recovery capacity of Italian ryegrass (determined as the rate of increase in the Nitrogen Nutrition Index during a rewatering period) was thrice that of tall fescue.

Nitrogen-use efficiency (biomass produced per unit of N absorbed), which is related to the final Nitrogen Nutrition Index obtained at harvest and which may reflect the integrated impact, is also affected by water shortage. In wheat, Fan and Li (2001) found that N-use efficiency was increased by water deficit and diminished by the dose of N applied. Pirmoradian et al. (2004) reached the same conclusions in rice, but the effect was only significant at high fertilizer application rates. In that case, and with a N application of 80 kg N  $\cdot$  ha<sup>-1</sup>, physiological efficiency (kg grain produced  $\cdot$  kg<sup>-1</sup> N removed) doubled when 50% more water was applied. Gajri et al. (1993) found a marked interaction between N and water for yield, and dependence of water-use efficiency on the N rate and N-use efficiency on the water supply. The effect of water deficit on plant N recovery capacity and on N-use efficiency was reviewed by Aulakh and Malhi (2005).

# 4 Effect of Water Deficit on Plant Nitrogen Demand

### 4.1 Growth of the Whole-Plant Biomass

It has been well established that water deficit reduces plant growth, primarily due to a reduction of the stomatal conductance that inhibits the C assimilation (Garwood et al., 1979; Bradford and Hsiao, 1982). Concerning yield, the stage at which drought occurs is critical; its effects are most pronounced if it takes place at an early stage. The first process affected by water deficit is foliar development and expansion (Fischer and Hagan, 1965; Bradford and Hsiao, 1982). As leaves are the vegetative organs that display higher N content during early developmental stages, N demand is also drastically reduced in early water deficits. Furthermore, during exponential growth, when the soil is not fully covered, drought restricts the active photosynthetic surface area, which causes a reduction in carbon assimilation and transpiration (Cowan, 1982; Durand et al., 1989) (see Fig. 1). When the leaf area index is below a value of approximately 3, the percentage of PAR absorbed to irradiated is below its potential maximum (Gosse et al., 1982; Durand et al., 1991; Akmal and Janssens, 2004). This also diminishes N needs (Nielsen and Halvorson, 1991), and may lead to a somewhat paradoxical situation where, for the same N uptake, an irrigated plant might experience N deficiency while a rainfed (and smaller) plant will maintain optimum nutrition status (Gonzalez-Dugo et al., 2005).

One source of complexity is that both N and water can affect leaf expansion and leaf photosynthesis simultaneously (Gastal and Saugier, 1989; Ghashghaie and Saugier, 1989, Durand et al., 1995; Jeuffroy et al., 2002) (Fig. 1), and their effects are intermingled. In tall fescue, experimental evidence suggested that water deficit did not alter the response of LER to N stress and vice versa (Durand, 1994). Under normal field conditions, such a reduction in photosynthesis is primarily a consequence of stomatal closure (Ghashghaie and Saugier, 1989). In the longer term, water deficit also induces N deficiency, that may further limit photosynthesis (Morgan, 1984; Lawlor et al., 1987; Ciompi et al., 1996; Arora et al., 2001). Photosynthetic rate is largely determined by the presence of RuBP carboxylase and chlorophyll content, both linearly related to leaf N content (Evans, 1989). Furthermore, Radin and Parker (1979) studying cotton plants and Broadley et al. (2001) studying lettuce demonstrated that soil N deficiency increased the sensitivity of stomata to water deficit, inducing lower leaf water potentials in a high transpiration regime, as measured in tall fescue (Onillon et al., 1995). Jacob et al. (1995) stated that the most important effect of N deficiency on plant water conductance occurs at the mesophyll level.

# 4.2 Root Shoot Ratio and Nitrogen Demand

The root/shoot ratio tends to increase with drought, largely due to a stronger effect of water deficit on shoot growth than on root growth (Sharp et al., 1988; Durand et al., 1989; Saab et al., 1990; Engels et al., 1994). In some cases, the root system may even continue to expand at very low water potential, whereas aboveground growth is completely halted (Wu et al., 1996). Because roots contain much lower concentrations of N than aboveground biomass, this change in the allocation pathway diminishes N needs. Measurements on roots are prone to major errors in the field (Pierret et al., 2005). Root mass changes under water deficit may be overestimated because of the restriction of root fragmentation and destruction by soil organisms, which depend on soil water content. However, some authors have found that the root growth of rice increased under water deficit (O'Toole, 1982; Ingram et al., 1994). An increase in root depth penetration has also been reported in rice (Mambani and Lal, 1983). By contrast, Yamauchi et al. (1996) observed a reduction in root growth as a result of drought. Engels et al. (1994) found that drought reduced root growth near the soil surface but enhanced it at lower levels. Onillon (1993) stated that the effect of water deficit on root dry

weight was dependent on N fertilizer application. In any case, such positive effects on root growth may not necessarily ensure improved conditions for N nutrition because newly colonized root zones are often deeper than N-rich topsoil horizons.

# 5 Effect of Water Deficit on Nitrogen Supply

# 5.1 Effect on Soil N Transformation Processes

Nitrogen transformation processes closely depend on water and its mobility in the soil. Among all the factors affecting soil nutrition, storage capacity and accessibility, soil texture, root depth and organic matter concentration deserve particular attention (Keller, 2005).

The size of soil  $NO_3^-$  and  $NH_4^+$  pools reflects the balance between various fluxes which result from several soil processes (Fig. 4): plant N uptake, mineralization, immobilization, nitrification and losses to the atmosphere following denitrification and volatilization (Scholes et al., 1997, Lemaire et al., 2004). Mineralization, immobilization and nitrification are affected by temperature and soil water content (Pastor and Post, 1985; Gorissen et al., 2004; Lemaire et al., 2004). Smolander et al. (2005) showed that a drought lasting less than two months was not sufficient to destroy microbial biomass; its activity was diminished but recovered completely soon after rewetting. Although sensitive to water deficits, N mineralization by



**Fig. 4** N cycle in the soil-plant-atmosphere system. Each process can be altered by the soil water regime. Dashed arrows show the processes that comprise net mineralization (After Lemaire et al., 2004)

soil micro-organisms may indeed be more resistant than is sometimes suggested, partly because immobilization could be more sensitive to water deficits (Valé et al., 2007).

The response of microbial biomass, nitrification and mineralization during dry periods has been studied in various types of vegetation (Mazzarino et al., 1998 in the Patagonian steppe; Pulleman and Tietema, 1999 in Pseudotsuga mensiezzii; Schimel et al., 1999 in the Alaskan taiga; Fierer and Schimel, 2002 in perennial oak and Bromus; Kelliher et al., 2004 in pine forest, Smolander et al., 2005 in a Norwegian Picea forest). Westerman and Tucker (1978) showed that during a wet period, the soil mineral N content decreased because of the increase in immobilization by microorganisms and losses to the atmosphere. Rain pulses can thus enhance microbial activity without exerting any significant effect on plant growth and hence on N demand by plants (White et al., 2004). During dry periods, root death increases (Huang and Gao, 2000); the soil organic labile N pool may do so as well (White et al., 2004). Austin et al. (2004) demonstrated that mineralization was activated by a water pulse after a drought period. However, following a dry period, during which root and bacterial death increased the amount of soil organic matter, both the immobilization rate and nitrification increased after soil rewetting (Smolander et al., 2005). No general effect of drought on long-term soil mineral N availability can thus be expected from such a complex network. Furthermore, because of the large error inherent in the measurements of root compartments, and given the importance of the disturbances caused by drought in cropping systems, this topic continues to be a major area for research.

# 5.2 Soil Fluxes: Transport of Nitrogen to the Roots, Mass Flow and Diffusion

Luxmoore and Millington (1971) indicated that the uptake of N and water may not occur throughout the entire root system. Using a system to inject nutrients at controlled depths in soil, the experiments carried out by Garwood and Williams (1967) proved that, as the soil became drier, N could be removed by a grass crop only when it was made available in deeper, moist soil horizons. Although other mechanisms might have been involved, their experiment was pioneering in suggesting that water flow influenced the local availability of N to roots. Indeed, insofar as N is dissolved in the soil solution, N uptake clearly depends on (i) water flows from the soil to the root system (Keller, 2005), and distribution of N and roots within the soil profile, and (ii) ion diffusion fluxes in the rhizosphere.

Water in the soil solution is loaded with anions, cations and soluble organic molecules, transported to the roots by the stream produced by the transpiration demand, creating convective flow or "mass flow" (Fig. 1). Thus, the amount of N solutes reaching the root surface is dependent on the water flux and the N concentration of the soil solution. Any trait in the soil or root system that alters water extraction from the soil will thus directly determine the associated flow of ions to the root surface. Kovacs (2005) demonstrated the importance of mass flow to the simulation of plant N uptake, and its effect on maize production.

Because root N uptake is an active process, if the plant N uptake capacity is higher than the amount of N reaching the root, a N concentration gradient can be built up in the rhizosphere, with lower concentrations at the root surface (Porporato et al., 2003). This causes a diffusive flow, expressed as follows:

$$\mathbf{J} = -\mathbf{D} \cdot \mathbf{A} \cdot \nabla \mathbf{c} \tag{5}$$

where J (mol·s<sup>-1</sup>) is the flux, and  $\nabla c$  (mol·cm<sup>-3</sup>·cm<sup>-1</sup>) the concentration gradient, D ( $cm^2 s^{-1}$ ) the diffusion coefficient and A (cm<sup>2</sup>) the area for diffusion. As already noted, measurements of A are known to display large errors (Pierret et al., 2005). A is generally estimated after root length density. The D value also (empirically) reflects the reduction in the diffusion rate due to chemical reactions, and the tortuousness of the trajectory, the latter being related to the spatial distribution of water-filled soil pores (Barber, 1974). Diffusion is thus strongly dependent on soil properties and humidity. It is also sensitive to dispersion, which depends on the shape and orientation of soil pores. This is generally expressed using an additional dispersion coefficient. In order to simulate the diffusive flow, the root can be considered as a cylinder of a given radius surrounded by a concentric soil volume (Gardner, 1960). Diffusion to a single root depends, in the first instance, on the radius of the concentric soil cylinder that is explored by the root, which is a function of the distance to the next root. This can be related to the root length density (Van Keulen, 1981; Klepper and Rickman, 1990).

When water flow is weak, or if the solution concentration is low, diffusion increases in relation to mass flow (Passioura, 1963; Raynaud, 2004). This is thus the case in low fertility media (Williams and Yanai, 1996) or with low water availability. Using a theoretical model (de Wit and van Keulen, 1972), Van Keulen (1981) performed a series of simulations to compare the effects of mass flow rate, dispersion and root density on anion extraction from the soil. He showed that the anion extraction rate increased markedly when root density was doubled or when dispersion was set at nil.

The theoretical framework was hence well established by all these authors. However, the leap to accurate determination of N flow under field conditions is difficult due to our limited knowledge of functional root architecture (Pierret et al., 2005). However, there have been some promising advances in the analysis of root architecture (Doussan et al., 2003; Pierret et al., 2007).

# 5.3 Plant Physiological Processes Involved in the Definition of Nitrogen Supply

To date, simulation models and experiments performed in nutrient solutions with labeled N have been the only techniques available to analyze N uptake independently of soil water dynamics. Caution should thus be adopted when drawing any quantitative conclusions using these approaches. There is, however, evidence of the direct effects of water deficit on N uptake. Nitrogen uptake is an active process. Based on an analysis of root respiration, Bloom et al. (1992) showed that the amount of energy required to absorb 1 mol  $NO_3^-$  was equivalent to 1 to 2 adenosine triphosphates (ATP), or 0.16 to 0.32 mol CO<sub>2</sub>. As for NH<sub>4</sub><sup>+</sup>, Bloom et al. (1992) estimated that 0.33 mol CO2 were needed to absorb and assimilate one mol of NH<sub>4</sub><sup>+</sup>. An independent assessment of nitrate influx and efflux has been possible since the 1980s through the use of stable isotope labeling. Several transport systems for nitrate uptake have been found, operating at two different levels of soil concentration (Siddiqi et al., 1990) and with different affinities for nitrate. The high-affinity transport

system is a low-concentration, saturable system. It reflects a strong ability for nitrate uptake at low concentrations in the soil solution, and it saturates between 0.2 and 0.5 mmol.m<sup>-3</sup> [NO<sub>3</sub><sup>-</sup>]. This high-affinity system is subject to negative feedback regulation by root NO<sub>3</sub> concentrations or certain products of NO<sub>3</sub> assimilation (Siddiqi et al., 1990). The low-affinity and nonsaturable transport system operates exclusively when the soil ion concentration is high, which may be a frequent occurrence under water deficit. This system has a linear relationship with soil nitrate concentrations and is only subject to negative feedback regulation after prolonged exposure to NO<sub>3</sub> (Siddiqi et al., 1990). It may be related to a passive movement through specific NO<sub>3</sub><sup>-</sup> channels, following an electrochemical potential gradient. Ammonium transport systems are similar to those of nitrate, as they also involve a lowcapacity, high-affinity system and a passive transport system with high capacity and low affinity.

Studies of the effects of plant-soil water relationships on root N uptake have produced contrasting results (BassiriRad et al., 1999). Nitrogen uptake can be considered as the minimum between N supply from soil and N demanded for growth. Some authors concluded that N uptake was independent of transpiration (Grubb, 1977; Schulze and Bloom, 1984; Gastal and Saugier, 1989; Hopmans and Bristow, 2002). Experiments analyzing the effect of the transpiration stream on N uptake during diurnal cycles reached similar conclusions, i.e., N uptake was found to be independent of transpiration (Triboi-Blondel, 1979; Bhat, 1982). During experiments carried out in nutrient solution with polyethylene glycol as an osmotic agent, Talouizite and Champigny (1988) indeed found an increase in nitrate uptake with short-term water deficit. Using <sup>15</sup>N-labeling, Matzner and Richards (1996) demonstrated that root capacity for N uptake diminished with a mild water deficit in Artemisia tridentata, but no further decrease was found when the drought stress was more severe. They stated that variations in the net effect of water deficit on N uptake might be correlated with the differing sensitivity of each transport system to water status. Buljovcic and Engels (2001) found a similar response in maize. In their experiment, the nitrate uptake capacity of excised maize roots fell to about 20% only when the soil water content decreased to 5% (w/w), corresponding to a soil water potential of about -3 MPa. During the same experiment, N uptake capacity was fully recovered after

rewatering. These authors concluded that nutrient uptake from dry soil was primarily regulated by nutrient transport in the soil to the root surface. Using split-root experiments with polyethylene glycol and isotopic labeling, Larsson (1992) showed that N uptake in wheat was strikingly reduced with osmotic stress. This effect seemed to be more closely related to the water status of the shoot than to the root osmotic environment, which may have been the reason for the different results of all these experiments. N accumulation in the roots may also cause the inhibition of nitrate uptake (Larsson, 1992; Gonzalez-Dugo, 2006).

Most of the physiological studies concerning the N uptake rate during water shortage periods have been done under controlled environments. Diouf et al. (2004) analyzed the effect of water deficit cycles on N uptake and N assimilation enzymes in the field by means of a plant N balance. There have been some attempts to assess the N uptake and allocation in the field, generally through the use of <sup>15</sup>N (Malagoli et al., 2005). However, the uncertainty about the size of the actual soil N pool due to N transformations under field conditions makes it difficult to make a quantitative assessment of N uptake by the plant, although it is a valuable tool for the analysis of the fertilizer recovery and the allocation pattern during plant ontogeny.

Thus, the processes that directly control N uptake under water deficit still need to be established, both at the molecular level (effect on transport system) and at the level of the whole plant.

# 6 Response of Crop Physiological Functions Involved in Nitrogen Nutrition to Water Deficit

### 6.1 Nitrogen Assimilation in Plant Tissues

Depending on the species, the assimilation of mineral N takes place in roots or shoots (Andrews, 1986), both exhibiting different energy costs. Gojon et al. (1994) suggested that the site of assimilation depended on the plant growth rate, with slow-growing species mainly assimilating via their roots, while fast-growing species assimilate nitrate in leaves. However, within a sample of eight herbaceous species exhibiting contrasted growth and assimilation rates, Scheurwater et al. (2002) demonstrated that assimilation systematically occurred in the leaves, irrespective of their growth rate.

The site of nitrate reduction may have an important effect on a plant's carbon budget. Plants that reduce nitrate in leaves could use the excess reducing power from photosynthesis, while those that reduce nitrate mainly in roots should obtain their reducing power from glycolysis and the oxidative pentose phosphate pathway (Scheurwater et al., 2002). As the energy cost of reducing nitrate in leaves is smaller than that in roots (Raven, 1985), in water deficit situations, plants that reduce nitrate in leaves are more efficient than those where the reduction takes place in roots.

Lawlor and Cornic (2002) stated that a water deficit could have a more marked effect on N assimilation than on the uptake process, as nitrate reductase activity diminished sharply, albeit in a reversible fashion, in line with low relative water content values. Other authors have also pointed out the particular sensitivity of nitrate reductase activity (Triboi-Blondel, 1978; Larsson et al., 1989; Larsson, 1992; Azedo-Silva et al., 2004; Correia et al., 2005).

# 6.2 Effect on Nitrogen Movement Through the Xylem

Dynamic variations in xylem sap composition result from differences in the transpiration rate and plantinternal nutrient relationships, which control the production of N compounds present in the xylem (Herdel et al., 2001; Peuke et al., 2001). Water shortage may thus have two different and opposite consequences regarding N flow. In the field, Bahrun et al. (2002) demonstrated that nitrate concentration in xylem sap in non-irrigated maize plants fell by more than 50% compared with irrigated control plants. Total nitrate flow was not measured in that experiment, but given the reduction in nitrate concentration and the decrease in stomatal conductance as a consequence of drought, it could be concluded that the net delivery of N to shoots was diminished. It is worth noting that in such a field experiment, the effect of water deficit on nitrate concentration is the result of cumulative effects on soil nitrate flow, nitrate uptake and the ratio between the decrease in nitrate and water uptake. The relationship between xylem flow and composition appears to be even more complex. It has been established that an increase in the solute concentration in xylem sap can reduce xylem hydraulic resistance (Zwieniecki et al., 2001). This would mean that for a given transpiration demand, water flow in the xylem may be enhanced if the solute concentration is increased. This is certainly valid when water availability is not limiting, but if plants are subjected to a water deficit and if an increase in solute concentration was the only consequence of a reduction in solvent volume, xylem sap flow would be diminished. In experiments performed in wheat, Larsson et al. (1989) showed that an increase in the osmotic pressure of nutrient solution increased xylem N concentration, although this increase did not compensate for the reduction in the transpiration rate. As a consequence, the delivery of nitrate to shoots decreased. By contrast, Nicolas et al. (1985) found in rain-fed wheat plants that the increase in N concentration in xylem sap compensated for the lower sap flux and that the amount of N reaching the shoots was maintained.

In sunflower plants grown under controlled conditions and in sandy loam-filled pots, Schurr and Schulze (1996) found that water deficit had no effect on xylem nitrate concentration. Indeed, under similar conditions, these levels even fell as the soil water content declined (Gollan et al., 1992). In that case, a restriction of N flow in the soil may have been involved. In another experiment in a controlled environment where transpiration and the nutrient supply were dissociated, Tanner and Beevers (2001) concluded that nutrient transport from roots to shoots was independent of transpiration. The two transpiration-independent water flows, i.e., the flux of water associated with volume expansion and Münch's counterflow in the phloem were found to ensure nutrient transport throughout the plant.

### 6.3 Remobilization of Nitrogen to Grains

For grain crops, N remobilization must also be taken into account, as most of the N that is present in grain has been taken up during the pre-anthesis period and thereafter remobilized from reserves or other organs. Grain number is set during flowering, so any limitation of crop growth rate during this period decreases grain number and yield in maize (Uhart and Andrade, 1995). Once grain number, and hence potential production, is established in the period from planting to seed set, carbon assimilation during seed filling and translocation of assimilates from reserves and senescing organs will set the final yield (Schnyder, 1993).

Nitrogen assimilated during the pre-anthesis period is the main source of nitrogenous compounds for grain filling; the contribution of reserves to the final grain N ranged from 63 to 100% in wheat, 11 to 100% in soybean, 49 to 64% in sorghum and 41 to 69% in maize (Egli, 2004; Barbottin et al., 2005), depending on their capacity to store large amounts of C and N compounds in their vegetative organs before anthesis. The amount of N remobilized depends on the amount of N available and on the N remobilization efficiency. The N remobilization efficiency is high in situations of low N supply or low N availability (Barbottin et al., 2005).

Water deficit hastens leaf senescence (Palta et al., 1994); maintenance of green leaf area is therefore essential to grain filling and yield under drought. This should also alter the C flow to roots in order to maintain N absorption (Dreccer, 2005). Palta et al. (1994) showed that N remobilization efficiency was high in Mediterranean-like conditions, where plants are subjected to drought stress during the grain-filling period.

# 7 Concepts Required for Simulation Models

Conceptual frameworks similar to that represented in Fig. 1 have been converted into a series of quantitative equations linked together to build crop models. Those simulation models are valuable tools to gather the knowledge acquired by experimentation. Mechanistic models enable the synthesis of many experimental results in order to achieve a global knowledge of plant function. They also allow the extrapolation of knowledge to different and future scenarios, a major requirement in the context of a continuously changing climate. Given the complexity of the interrelated processes considered here, models are relevant tools to discriminate between water- and N-related reductions in crop growth and development.

There is a large number of simulation models, appropriate for very different scales. Only some of the most used crop simulation models operating on a daily time step will be considered here to illustrate the different approaches to the effect of water and N interactions on crop growth and development. In modeling, deficits are generally defined by means of reducing indices that go from 0 to 1. The definition of these indices results from empirical equations relating state variables that are affected by the deficit. Each model thus exhibits a choice of significant state variables and equations defining the indices.

In the models considered here, the first effect of soil or plant water status on N nutrition is found at the soil level (Stockle et al., 1994; O'Leary and Connor, 1996; Van Ittersum et al., 2003; Singh et al., 2008). The amount of N that is available for plant uptake is calculated using the layers' water content and their contribution to total transpiration (Van Ittersum et al., 2003).

For most models, more than one index related to a deficit are used for different processes with a contrasted sensitivity to constraints. CROPSIM-WHEAT, CROPSYST and the model developed by O'Leary and Connor define the water deficit using the ratio between actual and potential transpiration (Hunt and Pararajasingham, 1995; Stockle et al., 1994; O'Leary and Connor, 1996). CERES-Maize and AFR-CWHEAT2 determine water deficit using a soil water balance (Jones and Kiriny, 1986; Porter, 1993). STICS considers the water available to roots above the permanent wilting point (Brisson et al., 2009).

Indices defining N deficit are generally calculated according to actual, minimum and maximum N content, in leaves, as in CROPSIM-WHEAT (Hunt and Pararajasingham, 1995) or in the aboveground biomass, as in AFRCWHEAT2 (Porter, 1993). STICS uses the Nitrogen Nutrition Index to assess N deficit (Brisson et al., 2009).

In general, biomass accumulation and leaf area formation are the processes that are typically noted as the most sensitive to these stresses. Some models use the most limiting factor, and therefore, the minimum between water and N stress indicator is used, underestimating the effect of the second stress. Other models consider that both processes interact in growth and development. Hence, both indices are considered. A summary of the most common models used, and how they account for water and N interaction in biomass development and leaf area formation, is given in Table 1. Singh et al. (2008) demonstrated that CropSyst was better at predicting the influence of water and N interaction on biomass and yield formation, compared with CERES-Wheat.

The relative difference between the two approaches depends on the values of both indices. Let us consider

Model	References	Biomass accumulation	LAI formation
Afrcwheat2	Porter, 1993	Water	Min (N, Water)
CERES-Maize	Jones and Kiriny, 1986	Min (N, Water)	Water
CropSim Wheat	Hunt and Pararajasingham, 1995	Min (N, Water)	Min (N, Water)
CropSyst	Stockle et al., 2003	Min (N, Water)	Water
Soygro	Sinclair, 1986	N*Water	N and water
STICS	Brisson et al., 2003	N*Water	Min (N, Water)
O'Leary and Connor	O'Leary and Connor, 1996	N*Water	N*Water

**Table 1** Revision of some daily time-step models and their interpretation of the effect of water and nitrogen interaction on biomass accumulation and leaf area formation



**Fig. 5** Difference (expressed by grayscale) between the minimum of water and nitrogen stress indices and the multiplication of both indices [Min (Water, N) – Water\*N] in the space defined by both indices

the result of the difference  $\varepsilon$  between the minimum of these indices and their multiplication:

$$\varepsilon = Min (w_{ater}, n_{itrogen}) - w_{ater} * n_{itrogen}$$

where  $w_{ater}$  and  $n_{itrogen}$  are stress factors of water and N, respectively.  $\varepsilon$  is largest when both indices are close to 0.5 (Fig. 5). When one of these indices is close to 0 (very limiting factor), the relative weight of this index becomes important and hence the effect of the second factor is weak. On the contrary, when one index is close to 1, the main effect on the variable considered will be mainly ascribed to the second factor. The maximal difference occurs in the intermediary region, i.e. when both indices are close to 0.5. Under natural conditions and because of its intrinsic inertia, N stress could be maintained close to this value. But in dry seasons, drought stress often remains at low values, reaching optimal conditions only following rainfalls. It is therefore maintained within these values for short periods of time only. As a result, the difference in the simulation results obtained using one or the other approach rarely appears significant.

### 8 Conclusion

The analysis of the effect of water deficit on N nutrition is complex and requires a multiscale approach, from the membrane to the crop, and different media and their interfaces. The first and most important need is to separate growth-induced changes from the direct effects of water deficits. The main effect of water restriction is certainly a reduction in N demand due to the marked sensitivity of leaf area expansion. Nitrogen nutrition indices can reveal the processes that alter plant nutrition independently of its actual demand. Processbased crop growth simulation models must currently incorporate these concepts, either implicitly or explicitly. It is noteworthy, however, that they still do not take similar account of the interaction between N and water stress, thus reflecting the ongoing discussion concerning co-limitation analysis in crops. Indeed, some aspects remain poorly understood, especially concerning the establishment of the supply of N to meet plant needs.

Two areas in particular deserve further investigative efforts. Firstly, our lack of knowledge of the true extension of active root surface area is strongly detrimental to any attempt to clarify the mechanisms of N absorption. Secondly, our inability to determine the mineral N absorption rate in the field still remains a major obstacle. To clarify these points, and render fertilizer application methods more precise, greater efforts should be made to achieve a methodological breakthrough in these two areas.

Acknowledgements The authors would like to thank Dr. Elías Fereres Castiel for his collaboration in the compilation of this paper. We also thank Dr. Jorge Alvarez de Toro. The anonymous reviewers are also acknowledged for their helpful comments that have improved this work. This research was supported by INRA, the Poitou-Charentes Region and the CNRS "ECosphère COntinentale" program, N#03CV114.

### References

- Addiscott T.M., Withmore A.P., Powlson D.S. (1991) Farming, fertilisers and the nitrate problem, CAB International, Wallingford, UK.
- Akmal M., Janssens M.J.J. (2004) Productivity and light use efficiency of perennial ryegrass with contrasting water and nitrogen supplies, Field Crop. Res. 88, 143–155.
- Alvarez de Toro J. (1987) Respueta del girasol (*Helianthus annuus* L.) a un suministro variable de agua de reigo y de nitrogeno, University of Cordoba.
- Andrews M. (1986) The partitioning of nitrate assimilation betwen root and shoot of higher plants, Plant Cell Environ. 9, 511–519.
- Angus J.F., Mancur M.W. (1985) Models of growth and development of wheat in relation to plant nitrogen, Aust. J. Agr. Res. 36, 537–544.
- Arora A., Singh V.P., Mohan. J. (2001) Effect of nitrogen and water stress on photosynthesis and nitrogen content in wheat, Biol. Plantarum 44, 153–155.
- Asseng S., Cao W., Zhang W., Ludwig F. (2009) Crop physiology, Modelling and Climate Change: Impact and Adaptation Strategies, in: Sadras V.O., Calderini D.F. (Eds.), Crop Physiology, Academic Press, Amsterdam, pp. 511–545.
- Aulakh M.S., Malhi S.S. (2005) Interactions of Nitrogen with Other Nutrients and Water: Effect on Crop Yield and Quality, Nutrient Use Efficiency, Carbon Sequestration, and Environmental Pollution, in: Sparks D.L. (Ed.), Advances in Agronomy, pp. 341–409.
- Austin A.T., Yahdjian L., Stark J.M., Belnap J., Porporato A., Norton U., Ravetta D.A., Schaffer S.M. (2004) Water pulses and biogeochemical cycles in arid and semiarid ecosystems, Oecologia 141, 221–235.
- Azedo-Silva J., Osório J., Fonseca F., Correia M.J. (2004) Effects of soil drying and subsequent re-watering on the activity of nitrate reductase in roots and leaves of *Helianthus annuus*, Funct. Plant Biol. 31, 611–621.
- Bahrun A., Jensen C.R., Asch F., Mogensen V.O. (2002) Drought-induced changes in xylem pH, ionic composition, and ABA concentration act as early signals in field-grown maize (*Zea mays L.*), J. Exp. Bot. 53, 251–263.
- Barber S.A. (1974) Influence of the plant root on ion movement in soils, in: Carson E.W. (Ed.), The Plant Root and its Environment, University Press of Virginia, Chalottesville, pp. 525–564.

- Barbottin A., Lecomte C., Bouchard C., Jeuffroy M.H (2005) Nitrogen Remobilization during Grain Filling in Wheat: Genotypic and Environmental Effects, Crop Sci. 45, 1141–1150.
- BassiriRad H., Tremmel D.C., Virginia R.A., Reynolds J.F., Soyza A.G.d., Brunell M.H. (1999) Short-term patterns in water and nitrogen acquisition by two desert shrubs following a simulated summer rain, Plant Ecol. 145, 27–36.
- Bélanger G., Walsh J.R., Richards J.E., Milburn P.H., Ziadi. N. (2001) Critical nitrogen curve and nitrogen nutrition index for potato in eastern Canada, Am. J. Potato Res. 78, 355– 364.
- Berni J.A.J., Zarco-Tejada P.J., Sepulcre-Cantó G., Fereres E., Villalobos F. (2009) Mapping canopy conductance and CWSI in olive orchards using high resolution thermal remote sensing imagery, Remote Sens. Environ. (to be published).
- Bhat K.K.S. (1982) Nutrient inflows into apple roots. II. Nitrate uptake rates measured on intact roots of mature trees under field conditions, Plant Cell Environ. 5, 461–469.
- Bloom A.J., Sukrapanna S.S., Warner R.L. (1992) Root respiration associated with ammonium and nitrate absorption and assimilation by Barley, Plant Physiol. 99, 1294–1301.
- Bradford K.J., Hsiao T.C. (1982) Physiological responses to moderate water stress, in: Lange O.L., Nobel P.S., Osmond C.B., Ziegler H. (Eds.), Physiological plant ecology II. Water realtions and carbon assimilation. Springer-Verlag, Berlin, pp. 263–324.
- Brisson N., Gary C., Justes E., Roche R., Mary B., Ripoche D., Zimmer D., Sierra J., Bertuzzi P., Burger P., Bussière F., Cabidoche Y.M., Cellier P., Debaeke P., Gaudillère J.P., Hénault C., Maraux F., Seguin B., Sinoquet H. (2003) An overview of the crop model STICS, Eur. J. Agron. 18, 309– 332.
- Brisson N., Launay M., Mary B., Beaudoin N. (2009) Conceptual basis, formalisations and parameterization of the STICS crop model, Quae, Versailles.
- Broadley M.R., A.J. Escobar-Gutierrez, Burns A., Burns I.G. (2001) Nitrogen-limited growth of lettuce is associated with lower stomatal conductance, New Phytol. 152, 97–106.
- Buljovcic Z., Engels C. (2001) Nitrate uptake ability by maize roots during and after drought stress, Plant Soil 229, 125–135.
- Cantero-Martinez C., Villar J.M., Romagosa I., Fereres E. (1995) Nitrogen fertilization of barley under semi-arid rainfed conditions, Eur. J. Agron. 4, 309–316.
- Cartelat A., Cerovic Z.G., Goulas Y., Meyer S., Lelarge C., Prioul J.L., Barbottin A., Jeuffroy M.H., Gate P., Agati G., Moya I. (2005) Optically assessed contents of leaf polyphenolics and chlorophyll as indicators of nitrogen deficiency in wheat (*Triticum aestivum* L.), Field Crop. Res. 91, 35–49.
- Cassman K.G., (2001) Science research to assure food security, in: Nösberger J., Geiger H.H., Struik P.C. (Eds.), Crop Science and Prospect, CABI Publishing, pp. 33–42.
- Celette F., Wery J., Chantelot E., Celette J., Gary C. (2005) Belowground interactions in a vine (*Vitis vinifera* L.)-tall fescue (*Festuca arundinacea Shreb.*) intercropping system: Water relations and growth, Plant Soil 276, 205–217.
- Ciompi S., Gentili E., Guidi L., Soldatini G.F. (1996) The effect of nitrogen deficiency on leaf gas exchange and chlorophyll fluorescence parameters in sunflower, Plant Sci. 118, 177–184.

- Colnenne C., Meynard J.M., Reau R., Justes E., Merrien A. (1998) Determination of a critical nitrogen dilution curve for winter oilseed rape, Ann. Bot. 81, 311–317.
- Correia M.J., Fonseca F., Azedo-Silva J., Dias C., David M.M., Barrote I., Osório M.L., Osório J., (2005) Effects of water deficit on the activity of nitrate reductase and content of sugars, nitrate and free amino acids in the leaves and roots of sunflower and white lupin plants growing under two nutrient supply regimes, Physiol. Plantarum 124, 61–70.
- Cowan I.R., (1982) Economics of carbon fixation in higher plants, in: Givnish T.J. (Ed.), On the economy of plant form and function, Cambridge University Press, Cambridge, pp. 133–171.
- De Wit C.T., Van Keulen H. (1972) Simulation of transport processes in soils, Centre for Agricultural Publishing and Documentation Wageningen, Wageningen, Netherlands.
- Devienne-Barret F., Justes E., Machet J.M., Mary B. (2000) Integrated control of nitrate uptake by crop growth rate and soil nitrate availability under field conditions, Ann. Bot. 86, 995–1005.
- Diouf O., Brou Y.C., Diouf M., Sarr B., Eyletters M., Roy-Macauley H., Delhaye J.P. (2004) Response of pearl millet to nitrogen as affected by water deficit, Agronomie 24, 77–84.
- Dixon H.H., Joly J. (1895) On the ascent of sap, Philos. T. Roy. Soc. 186, 563–576.
- Doussan C., Pagés L., Pierret A. (2003) Soil exploration and resource acquisition by plant roots: An architectural and modelling point of view, Agronomie 23, 419–431.
- Dreccer M.F. (2005) Nitrogen use at the leaf and canopy level: a framework to improve crop N use efficiency, J. Crop Improv. 15, 97–125.
- Durand J.L. (1994) Response of morphogenesis to water deficits and competition, in: Sinoquet H., Cruz P. (Eds.), Ecophysiology of tropical intercropping, INRA, Paris, pp. 257–274.
- Durand J.L., Lemaire G., Gosse G., Chartier M. (1989) Analyse de la conversion de l'énergie solaire en maitère sèche par un peuplement de luzene (*Medicago sativa* L.) soumis à un déficit hydrique, Agronomie 9, 599–607.
- Durand J.L., Onillon B., Schnyder H., Rademacher I. (1995) Drought effects on cellular and spatial parameters of leaf growth in tall fescue, J. Exp. Bot. 46, 1147–1155.
- Durand J.L., Varlet-Grancher C., Lemaire G., Gastal F., Moulia B. (1991) Carbon partitioning in forage crops, Acta Biotheor. 39, 213–224.
- Duru M. (2004) Simplified nitrogen assessment of orchardgrass swards, Agron. J. 96, 1598–1605.
- Easterling D.R., Meehl G.A., Parmesan C., Changnon S.A, Karl T.R., Mearns L.O. (2000) Climate extremes: Observations, Modeling, and Impacts, Science 289, 2068–2074.
- Egli D.B. (2004) Seed-fill duration and yield of grain crops, Adv. Agron. 83, 243–279.
- Engels C., Mollenkopf M., Marschner H. (1994) Effect of drying and rewetting the topsoil on root growth of maize and rape in different soil types, Z. Pflanzenernähr. Bodenk. 157, 139– 144.
- Evans J.R. (1989) Photosynthesis and nitrogen relationships in leaves of C3 plants, Oecologia 78, 9–19.
- Fan X.L., Li Y.K. (2001) Effect of drought stress and drought tolerancec heredity on nitrogen efficiency of winter wheat, in: Horst W.W.J., Schenk M.K., Burkert A., Claasen N.,

Flessa H., Frommer W.B., Goldbach H.E., Olfs H.-W., Romheld W., Sattelmacher B., Schmidhalter U., Schubert S., von Wiren N., Wittenmayer L. (Eds.), Plant Nutrition: Food Security and Sustainability of Agro-Ecosystems, American Society of Agronomy, Madison, USA, pp. 62–63.

- Farruggia A., Gastal F., Scholefield D. (2004) Assessment of the nitrogen status of grassland, Grass Forage Sci. 59, 113–120.
- Field, C., Mooney H.A. (1986) The photosynthesis-nitrogen relationship in wild plants, in: Givnish T.J. (Ed.), On the economy of plant form and function, Cambridge University Press, Cambridge, pp. 25–55.
- Fierer N., Schimel J.P. (2002) Effects of drying-rewetting frequency on soil carbon and nitrogen transformations, Soil Biol. Biochem. 34, 777–787.
- Fischer R.A., Hagan R.M. (1965) Plant water relations, irrigation management and crop yield, Exp. Agr. 1, 161–177.
- Gajri P.R., Prihar S.S., Arora V.K. (1993) Interdependence of nitrogen and irrigation effects on growth and input-use efficiencies in wheat, Field Crop. Res. 31, 71–86.
- Galloway J.N., Townsend A.R., Erisman J.W., Bekunda M., Cai Z., Freney J.R., Martinelli L.A., Seitzinger S.P., Sutton M.A. (2008) Transformation of the Nitrogen Cycle: Recent Trends, Questions, and Potential Solutions, Science 320, 889–892.
- Gamon J.A., Peñuelas J., Field C.B. (1992) A narrow-wave band spectral index that track diurnal changes in photosynthetic efficiency, Remote Sens. Environ. 41, 35–44.
- Gardner W.R. (1960) Dynamic aspects of water availability to plants, Soil Sci. 89, 63–73.
- Garwood E.A., Tyson K.C., Sinclair J. (1979) Use of water by six grass species. 1. Dry-matter yields and response to irrigation, J. Agr. Sci. 93, 13–24.
- Garwood E.A., Williams T.E. (1967) Growth, water use and nutrient uptake from the subsoil by grass swards, J. Agr. Sci. 69, 125–130.
- Gastal F., Lemaire G. (2002) N uptake and distribution in crops: an agronomical and ecophysiological perspective, J. Exp. Bot. 53, 789–799.
- Gastal F., Saugier B. (1989) Relationships between nitrogen uptake and carbon assimilation in whole plants of tall fescue, Plant Cell Environ. 12, 407–418.
- Ghashghaie J., Saugier B. (1989) Effects of nitrogen deficiency on leaf photosynthetic response of tall fescue to water deficit, Plant Cell Environ. 12, 261–271.
- Giorgi F., Bi X. (2005) Regional changes in surface climate interannual variability for the 21th century from ensembles of global model simulations, Geophys. Res. Lett. 32.
- Glass A.D.M., Britto D.T., Kaiser B.N., Kinghorn J.R., Kronzucker H.J., Kumar A., Okamoto M., Rawat S., Siddiqi M.Y., Unkles S.E., Vidmar J.J. (2002) The regulation of nitrate and ammonium transport systems in plants, J. Exp. Bot. 53, 855–864.
- Gojon A., Passard C., Bussi C. (1994) Root/shoot distribution of NO<sub>3</sub><sup>-</sup> assimilation in herbaceous and woody species, in: Roy R., Garnier E. (Eds.), A whole plant perspective on carbon-nitrogen interactions, SPB Academic Publishing, The Hague.
- Gollan T., Schurr U., Schulze E.D. (1992) Stomatal response to drying soil in relation to changes in the xylem sap composition of *Helianthus annuus* I. The concentration of cations anions, amino acids in, and pH of, the xylem sap, Plant Cell Environ. 15, 551–559.
- Gonzalez-Dugo V. (2006) Effet du déficit hydrique sur l'état de nutrition azotée chez les graminées fourragères, University of Poitiers, France, 189.
- Gonzalez-Dugo V., Durand J.L., Gastal F., Picon-Cochard C. (2005) Short-term response of the nitrogen nutrition status of tall fescue and Italian ryegrass swards under water deficit, Aust. J. Agr. Res. 56, 1269–1276.
- Gorissen A., Tietema A., Joosten N.N., Estiarte M., Peñuelas J., Sowerby A., Emmett B.A., Beier C. (2004) Climate change affects carbon allocation to the soil in shrublands, Ecosystems 7, 650–661.
- Gosse G., Chartier M., Varlet-Granchet C., Bonhomme R. (1982) Interception du rayonnement utile à la photosynthèse chez la luzerne : Variations et modélisation, Agronomie 2, 583–588.
- Greenwood D.J., Lemaire G., Gosse G., Cruz P., Draycott A., Neeteson J.J. (1990) Decline in percentage N of C3 and C4 crops with increasing plant mass, Ann. Bot. 66, 425–436.
- Grubb P.J. (1977) Control of forest growth and distribution on wet tropical mountains: with special reference to mineral nutrition, Ann. Rev. Ecol. Syst. 8, 83–107.
- Hardwick R.C. (1987) The nitrogen content of plants and the self-thinning rule of plant ecology: A test of the core-skin hypothesis, Ann. Bot. 60, 439–446.
- Herdel K., Schmidt P., Feil R., Mohr A., Schurr U. (2001) Dynamics of concentrations and nutrient fluxes in the xylem of *Ricinus cummunis*-diurnal course, impact of nutrient availability and nutrient uptake, Plant Cell Environ. 24, 41–52.
- Hikosaka K., Terashima I., Katoh S. (1994) Effects of leaf age, nitrogen nutrition and photon flux density on the distribution of nitrogen among leaves of a vine (*Ipomoea tricolor* Cav.) grown horizontally to avoid mutual shading of leaves, Oecologia 97, 451–457.
- Hirose T., Werger M.J.A. (1987) Maximizing daily canopy photosynthesis with respect to the leaf nitrogen allocation pattern in the canopy, Oecologia 72, 520–526.
- Hopmans J.W., Bristow K.L. (2002) Current capabilities and future needs of root water and nutrient uptake modeling, Adv. Agron. 77, 103–183.
- Huang B., Gao H. (2000) Root physiological characteristics associated with drought resistance in tall fescue cultivars, Crop Sci. 40, 196–203.
- Hunt L.A., Pararajasingham S. (1995) CROPSIM WHEAT: A model describing the growth and development of wheat, Can. J. Plant Sci. 75 619–632.
- Idso S., Jackson R., Pinter P.J.J., Reginato R., Hatfield J. (1981) Normalizing the stressdegree-day parameter for environmental variability, Agr. Meteorol. 24, 45–55.
- Ingram K.T., Bueno F.D., Namuco O.S., Yambao E.B., Beyrouty C.A. (1994) Rice root traits for drought resistance and their genetic variation, Rice Roots: Nutrient and Water Use, 67– 77.
- Jackson R., Idso S., Reginato R., Pinter P.J. (1981) Canopy temperature as a crop water stress indicator, Water Resour. Res. 17, 1133–1138.
- Jacob J., Udayakumar M., Prasad T.G. (1995) Mesophyll conductance was inhibited more than stomatal conductance in nitrogen deficient plants, Plant Physiol. Bioch. 17, 55–61.

- Jeuffroy M.H., Ney B., Ourry A. (2002) Integrated physiological and agronomic modelling of N capture and use within the plant, J. Exp. Bot. 53, 809–823.
- Jones C.A., Kiriny R. (1986) CERES-Maize: A Simulation Model of Maize Growth and Development, Texas A & M University Press, College Station.
- Jury W.A., Vaux J.H. (2005) The role of science in solving the world's emerging water problems, PNAS 102, 15715–15720.
- Justes E., Mary B., Meynard J.M., Machet J.M., Thelier-Huche L. (1994) Determination of a critical nitrogen dilution curve for winter wheat crops, Ann. Bot. 74, 397–407.
- Keller M. (2005) Deficit irrigation and vine mineral nutrition, Am. J. Enol. Viticult. 56, 267–283.
- Kelliher F.M., Ross D.J., Law B.E., Baldocchi D.D., Rodda N.J. (2004) Limitations to carbon mineralization in litter and mineral soil of young and old ponderosa pine forests, Forest Ecol. Manag. 191, 201–213.
- Klepper B., Rickman R.W. (1990) Modeling crop growth and function, Adv. Agron. 44, 113–132.
- Kovács G.J. (2005) Modelling of adaptation processes of crops to water and nitrogen stress, Phys. Chem. Earth 30, 209–216.
- Larsson M. (1992) Translocation of nitrogen in osmotically stressed wheat seedlings, Plant Cell Environ. 15, 447–453.
- Larsson M., Larsson C.M., Whitford P.N., Clarkson D.T. (1989) Influence of osmotic stress on nitrate reductase activity in wheat (*Triticum aestivum* L.) and the role of abscisic acid, J. Exp. Bot. 40, 1265–1271.
- Lawlor D.W., Cornic G. (2002) Photosynthetic carbon assimilation and associated metabolism in relation to water deficits in higher plants, Plant Cell Environ. 25, 275–294.
- Lawlor D.W., Young A.T., Keys A.J., Kendall A.C. (1987) Nitrate nutrition and temperature effects on wheat: photosynthesis and photorespiration of leaves, J. Exp. Bot. 38, 378– 392.
- Lemaire G., Denoix A. (1987) Croissance estivale en matière sèche de peuplements de fétuque élevée (*Festuca arundinacea* Schreb.) et de dactyle (*Dactylis glomerata* L.) dans l'Ouest de la France. II. Interaction entre les niveaux d'alimentation hydrique et de nutrition azotée, Agronomie 7, 381–389.
- Lemaire G., Gastal F. (1997) N uptake and distribution in plant canopies, in: Lemaire G. (Ed.), Diagnosis on the nitrogen status in crops, Springer-Verlag, Heidelberg, pp. 3–43.
- Lemaire G., Gastal F. (2009) Quantifying Crop Responses to Nitrogen Deficiency and Avenues to improve Nitrogen Use Efficiency, in: Sadras V.O., Calderini D.F. (Eds.), Crop Physiology, Academic Press, Amsterdam, pp. 171–212.
- Lemaire G., Meynard J.M. (1997) Use of the nitrogen nutrition index for the analysis of agronomical data, Diagnosis of the Nitrogen Status in Crops, 45–55.
- Lemaire G., Salette J. (1984) Rélation entre dynamique de croissance et dynamique de prélèvement d'azote pour un peuplement de graminées fourragères. I. Étude de l'effet du milieu, Agronomie 4, 423–430.
- Lemaire G., Recous S., Mary B. (2004) Managing residues and nitrogen in intensive cropping systems. New understanding for efficient recovery by crops, in:. Fischer T., Turner N.C., Angus J.F., McIntyre L., Robertson M.J., Borrell A., Lloyd D. (Eds.), 4th International Crop Science Congress, The

Regional Institute Ltd, Gosford, Brisbane (Australia), http:// www.cropscience.org.au/icsc2004.

- Li S.X., Wang Z.H., Hu T.T., Gao Y.J., Stewart B.A. (2009) Chapter 3. Nitrogen in Dryland Soils of China and Its Management, in: Sparks D.L. (Ed.), Advances in Agronomy, pp. 123–181.
- Luxmoore R.J., Millington R.J. (1971) Growth of perennial ryegrass (*Lolium perenne* L.) in relation to water, nitrogen, and light intensity - II. Effects on dry weight production, transpiration and nitrogen uptake, Plant Soil 34, 561–574.
- Macduff J.H., Jarvis S.C., Mosquera A. (1989) Nitrate nutrition of grasses from steady-state supplies in flowing solution culture following nitrate deprivation and/or defoliation.
  II. Assimilation of NO<sub>3</sub> and short-term effects on NO<sub>3</sub> uptake, J. Exp. Bot. 40, 977–984.
- Malagoli P., Laine P., Rossato L., Ourry. A. (2005) Dynamics of nitrogen uptake and mobilization in field-grown winter oilseed rape (*Brassica napus*) from stem extension to harvest: I. Global N flows between vegetative and reproductive tissues in relation to leaf fall and their residual N, Ann. Bot. 95, 853–861.
- Mambani B., Lal R. (1983) Response of upland rice varieties to drought stress - II. Screening rice varieties by means of variable moisture regimes along a toposequence, Plant Soil 73, 73–94.
- Matzner S.L., Richards J.H. (1996) Sagebrush (Artemisia tridentata Nutt.) roots maintain nutrient uptake capacity under water stress, J. Exp. Bot. 47, 1045–1056.
- Mazzarino M.J., Bertiller M.B., Sain C., Satti P., Coronato F. (1998) Soil nitrogen dynamics in northeastern Patagonia steppe under different precipitation regimes, Plant Soil 202, 125–131.
- Mistele B., Schmidhalter U. (2008) Estimating the nitrogen nutrition index using spectral anatomy reflectance measurements, Eur. J. Agron. 29, 184–190.
- Morgan J.A. (1984) Interaction of water supply and nitrogen in wheat, Plant Physiol. 76, 112–117.
- Ney B., Doré T., Sagan M. (1997) The nitrogen requirement of major agricultural crops: Grain legumes, Diagnosis of the Nitrogen Status in Crops, 107–118.
- Nicolas M.E., Simpson R.J., Lambers H., Dalling M.J. (1985) Effects of drought on partitioning of nitrogen in two wheat varieties differing in drought tolerance, Ann. Bot. 55, 743–754.
- Nielsen D.C., Halvorson A.D. (1991) Nitrogen fertility influence on water stress and yield of winter wheat, Agron. J. 83, 1065–1070.
- O'Leary G.J., Connor D.J. (1996) A simulation model of the wheat crop in response to water and nitrogen supply: I. Model construction, Agr. Syst. 52, 1–29.
- O'Toole J.C. (1982) Adaptation of rice to drought-prone environments. Drought resistance in crops with the emphasis on rice, IRRI, Manila, pp. 71–82.
- Onillon B. (1993) Effets d'une contrainte hydrique édaphique sur la croissance de la fétuque élevée soumise à différents niveaux de nutrition azotée. Étude à l'echelle foliaire et à celle du couvert végétal, University of Poitiers.
- Onillon B., Durand J.L., Gastal F., Tournebize R. (1995) Drought effects on growth and carbon partitioning in a tall fescue sward grown at different rates of nitrogen fertilization, Eur. J. Agron. 4, 91–99.

- Overman A.R., Robinson D., Wilkinson S.R. (1995) Coupling of dry matter and nitrogen accumulation in ryegrass, Fertil. Res. 40, 105–108.
- Palta J.A., Kobata T., Turner N.C., Fillery I.R. (1994) Remobilization of Carbon and Nitrogen in Wheat as Influenced by Postanthesis Water Deficits, Crop Sci. 34, 118–124.
- Passioura J. (1963) A mathematical model for the uptake of ions from the soil solution, Plant Soil 18, 225–238.
- Pastor J., Post W.M. (1985) Development of a linked forest productivity-soil process model, Oak Ridge National Laboratory.
- Peuke A.D., Rokitta M., Zimmermann U., Schreiber L., Haase A. (2001) Simultaneous measurement of water flow velocity and solute transport in xilem and phloem of adult plants of *Ricinus communis* over a daily time course by nuclear magnetic resonance spectrometry, Plant, Cell Environ. 24, 491–503.
- Phillip J.R. (1966) Plant water relations: some physical aspects, Annu. Rev. Plant Phys. 17, 245–268.
- Pierret A., Doussan C., Capowiez Y., Bastardie F., Pagès L. (2007) Root functional architecture: A framework for modeling the interplay between roots and soil, Vadose Zone Journal 6, 269–281.
- Pierret A., Moran C.J., Doussan C. (2005) Conventional detection methodology is limiting our ability to understand the roles and functions of fine roots, New Phytol. 166, 967–980.
- Pirmoradian N., Sepaskhah A.R., Maftoun M. (2004) Deficit irrigation and nitrogen effects on nitrogen-use efficiency and grain protein of rice, Agronomie 24, 143–153.
- Plénet D., Lemaire G. (1999) Relationships between dynamics of nitrogen uptake and dry matter accumulation in maize crops, Determination of critical N concentration, Plant Soil 216, 65–82.
- Porporato A., D'Odorico P., Laio F., Rodriguez-Iturbe I. (2003) Hydrologic controls on soil carbon and nitrogen cycles I. Modeling scheme, Adv. Water Resour. 26, 45–58.
- Porter J.R. (1993) AFRCWHEAT2: a model of the growth and development of wheat incorporating responses to water and nitrogen, Eur. J. Agron. 2, 69–82.
- Pulleman M., Tietema A. (1999) Microbial C and N transformations during drying and rewetting of coniferous forest floor material, Soil Biol. Biochem. 31, 275–285.
- Radin J.W., Parker L.L. (1979) Water relations of cotton plants under nitrogen deficiency. II. Environmental interactions on stomata, Plant Physiol. 64, 499–501.
- Rao K.P., Rains D.W. (1976) Nitrate absorption by barley. I. Kinetics and energetics, Plant Physiol. 57, 55–58.
- Raven J.A. (1985) Regulation of pH and generation of osmolarity in vascular plants: a cost-benefit analysis in relation to efficiency of use of energy, nitrogen and water, New Phytol. 101, 25–77.
- Raynaud X. (2004) Interactions compétitives spatialisées dans le sols : une approche par modélisation des interactions entre les plantes et les micro-organismes du sol, Université de Paris-Sud.
- Rodriguez D., Robson A.J., Belford R. (2009) Dynamic and functional monitoring technologies for applications in crop management, in: Sadras V.O., Calderini D.F. (Eds.), Crop Physiology, Academic Press, Amsterdam, pp. 489–510.

- Saab I.N., Sharp R.E., Pritchard J., Voetberg G.S. (1990) Increased endogenous abscisic acid maintains primary root growth and inhibits shoot growth of maize seedlings at low water potentials, Plant Physiol. 93, 1329–1336.
- Sadras V.O. (2005) A quantitative top-down view of interactions between stresses: theory and analysis of nitrogen-water colimitation in Mediterranean agro-ecosystems, Aust. J. Agr. Res. 56, 1151–1157.
- Scheurwater I., Koren M., Lambers H., Atkin O.K. (2002) The contribution of roots and shoots to whole plant nitrate reduction in fast-and slow-growing grass species, J. Exp. Bot. 53, 1635–1642.
- Schimel J.P., Gulledge J.M., Clein-Curley J.S., Lindstrom J.E., Braddock J.F. (1999) Moisture effects on microbial activity and community structure in decomposing birch litter in the Alaskan taiga, Soil Biol. Biochem. 31, 831–838.
- Schnyder H. (1993) The role of carbohydrate storage and redistribution in the source - sink relations of wheat and barley during grain filling - a review, New Phytol. 123, 233–245.
- Scholes M.C., Martin R., Scholes R.J., Parsons D., Winstead E. (1997) NO and N<sub>2</sub>O emissions from savanna soils following the first simulated rains of the season, Nutr. Cycl. Agroecosys. 48, 115–122.
- Schulze E.D., Bloom A.J. (1984) Relationship between mineral nitrogen influx and transpiration in radish and tomato, Plant Physiol. 76, 827–828.
- Schurr U., Schulze E.D. (1996) Effects of drought on nutrient and ABA transport in *Ricinus communis*, Plant Cell Environ. 19, 665–674.
- Sharp R.E., Silk W.K., Hsiao T.C. (1988) Growth of the primary root at low water potentials, Plant Physiol. 87, 50–57.
- Sheehy J. (2001) Will yield barriers limit future rice production? in: Nösberger J., Geiger H.H., Struik P.C. (Eds.), Crop Science and Prospect, CABI Publishing, pp. 281–306.
- Siddiqi M.Y., Glass A.D.M., Ruth T.J., Rufti J.T.W. (1990) Studies of the uptake of nitrate in Barley, Plant Physiol. 93, 1426–1432.
- Sinclair T.R. (1986) Water and nitrogen limitations in soybean grain production. I. Model development, Field Crop. Res. 15, 125–141.
- Singh A.K., Tripathy R., Chopra U.K. (2008) Evaluation of CERES-Wheat and CropSyst models for water-nitrogen interactions in wheat crop, Agr. Water Manage. 95, 776–786.
- Smolander A., Barnette L., Kitunen V., Lumme I. (2005) N and C transformations in long-term N-fertilized forest soils in response to seasonal drought, Appl. Soil Ecol. 29, 225–235.
- Stevens C.J., Dise N.B., Mountford J.O., Gowing D.J. (2004) Impact of Nitrogen Deposition on the Species Richness of Grasslands, Science 303, 1876–1879.
- Stöckle C.O., Donatelli M., Nelson R. (2003) CropSyst, a cropping systems simulation model, Eur. J. Agron. 18, 289–307.
- Stöckle C.O., Martin S.A., Campbell G.S. (1994) CropSyst, a cropping system simulation model: water/nitrogen budgets and crop yield, Agr. Syst. 46, 335–359.
- Streeter J.G. (2003) Effects of drought on nitrogen fixation in soybean root nodules, Plant Cell Environ. 26, 1199–1204.

- Suárez L., Zarco-Tejada P.J., Berni J.A.J., Gonzalez-Dugo V., Fereres E. (2009) Modelling PRI for water stress detection using radiative transfer models, Remote Sens. Environ. 113, 730–744.
- Talouizite A., Champigny M.L. (1988) Response of wheat seedlings to short-term drought stress with particular respect to nitrate utilisation, Plant Cell Environ. 11, 149–155.
- Tanner W., Beevers H. (2001) Transpiration, a prerequisite for long-distance transport of minerals in plants? PNAS 98, 9443–9447.
- Thomas M., Robertson M.J., Fukai S., Peoples M.B. (2004) The effect of timing and severity of water deficit on growth, development, yield accumulation and nitrogen fixation of mungbean, Field Crop. Res. 86, 67–80.
- Tischner R. (2000) Nitrate uptake and reduction in higher and lower plants. Plant Cell Environ. 23, 1005–1024.
- Triboi-Blondel A.-M. (1978) Effets de différents régimes d'alimentation hydrique sur l'activité in vivo de la nitratereductase, C.R. Acad. Sci. Paris 286, 1795–1798.
- Triboi-Blondel A.-M. (1979) Dynamique comparée de l'absorption des nitrates et de l'eau par des plantules de blé, C.R. Acad. Sci. Paris 288, 1545–1548.
- Tyree M.T., Cochard H. (2003) Vessel contents of leaves after excision: a test of the scholander assumption, J. Exp. Bot. 54, 2133–2139.
- Uhart S.A., Andrade F.H. (1995) Nitrogen Defeciency in Maize: I. Effects on Crop Growth, Development, Dry Matter Partitioning, and Kernel Set, Crop Sci. 35, 1376–1383.
- Valé M., Mary B., Justes E. (2007) Irrigation practices may affect denitrification more than nitrogen mineralization in warm climatic conditions, Biol. Fert. Soils 43, 641–651.
- Van Der Honert T.H. (1948) Water transport as a catenary process, Discuss. Faraday Soc. 3, 146–153.
- Van Dobben W.H. (1962) Influence of temperature and light conditions on dry-matter distribution, develoment rate and yield in arable crops, Neth. J. Agr. Sci. 10, 377–389.
- Van Ittersum M.K., Leffelaar P.A., Van Keulen H., Kropff M.J., Bastiaans L., Goudriaan J. (2003) On approaches and applications of the Wageningen crop models, Eur. J. Agron. 18, 201–234.
- Van Keulen H. (1981) Modelling the interaction of water and nitrogen, Plant Soil 58, 205–229.
- Van Keulen H., Seligman N.G. (1987) Simulation of water use, nitrogen nutrition and growth of a spring wheat crop, Pudoc, Wageningen.
- Wei C., Steudle E., Tyree M.T. (1999) Water ascent in plants: Do ongoing controversies have a sound basis? Trends Plant Sci. 4, 372–375.
- Westerman R.L., Tucker T.C. (1978) Denitrification in desert soils, Nitrogen in Desert Ecosystems, 75–106.
- White C.S., Moore D.I., Craig J.A. (2004) Regional-scale drought increases potential soil fertility in semiarid grasslands, Biol. Fert. Soils 40, 73–78.
- Williams M., Yanai R.D. (1996) Multi-dimensional sensitivity analysis and ecological implications of a nutrient uptake model, Plant Soil 180, 311–324.
- Wu Y., Sharp R.E., Durachko D.M., Cosgrove D.J. (1996) Growth Maintenance of the Maize Primary Root at Low

Water Potentials Involves Increases in Cell-Wall Extension Properties, Expansin Activity, and Wall Susceptibility to Expansins, Plant Physiol. 111, 765–772.

Yamauchi A., Paradales Jr J.R., Kono Y. (1996) Root system structure and its relation to stress tolerance, in: Ito O., Katayama K., Johansen J.V.D.K., Kumar Rao J.J. (Eds.), Root and nitrogen in cropping systems of the semiarid tropics, Cultio Corporation, Tsukuba, Japan, pp. 211–233.

Zwieniecki M.A., Melcher P.J., Holbrook. N.M. (2001) Hidrogel control of xilem hydraulic resistance in plants, Science 291, 1059–1062.

# Validation of Biophysical Models: Issues and Methodologies

Gianni Bellocchi, Mike Rivington, Marcello Donatelli, and Keith Matthews

## Contents

1	<b>Introduction</b> 5				
2	Issues	on Model Validation	581		
	2.1	Validation Purpose	581		
	2.2	Interpretation of Phenomena	582		
	2.3	Model Comparison	583		
	2.4	Model Predictions	583		
	2.5	Model Complexity	584		
	2.6	Data Accuracy and Quality	585		
	2.7	Robustness of Model Results	587		
	2.8	Time Histories	587		
	2.9	Summary	587		
3	Valida	tion of Models	588		
	3.1	Validation Measures	588		
	3.2	Disaggregating Statistics	590		
	3.3	Statistical Hypothesis Tests	591		
	3.4	Validation Criteria	593		
	3.5	Combining Multiple Statistics	593		
	3.6	Summary	595		
4	Concl	usion	596		
Re	<b>References</b>				

Abstract The potential of mathematical models is widely acknowledged for examining components and interactions of natural systems, estimating the changes and uncertainties on outcomes, and fostering communication between scientists with different backgrounds and between scientists, managers and the community. For favourable reception of models, a systematic accrual of a good knowledge base is crucial for both science and decision-making. As the roles of models grow in importance, there is an increase in the need for appropriate methods with which to test their quality and performance. For biophysical models, the heterogeneity of data and the range of factors influencing usefulness of their outputs often make it difficult for full analysis and assessment. As a result, modelling studies in the domain of natural sciences often lack elements of good modelling practice related to model validation, that is correspondence of models to its intended purpose. Here we review validation issues and methods currently available for assessing the quality of biophysical models. The review covers issues of validation purpose, the robustness of model results, data quality, model prediction and model complexity. The importance of assessing input data quality and interpretation of phenomena is also addressed. Details are then provided on the range of measures commonly used for validation. Requirements for a methodology for assessment during the entire model-cycle are synthesised. Examples are used from a variety of modelling studies which mainly include agronomic modelling, e.g. crop growth and development, climatic modelling, e.g. climate scenarios, and hydrological modelling, e.g. soil hydrology, but the principles are essentially applicable to any area. It is shown that conducting detailed validation requires multi-faceted knowledge, and poses

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_26,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Bellocchi et al., Agron. Sustain. Dev. 30 (2010) 109–130. DOI: 10.1051/agro:2009001

G. Bellocchi (🖂)

Grassland Ecosystem Research Unit, French National Institute for Agricultural Research, 234 Avenue du Brézet, 63100 Clermont-Ferrand, France e-mail: gianni.bellocchi@clermont.inra.fr, giannibellocchi@yahoo.com

substantial scientific and technical challenges. Special emphasis is placed on using combined multiple statistics to expand our horizons in validation whilst also tailoring the validation requirements to the specific objectives of the application.

**Keywords** Accuracy • Modelling • Multiple statistics • Validation

#### Abbreviations

AIC	Akaike's information criterion
BIC	Bayesian information criterion
CD	Coefficient of determination
$C_p$	Mallows' statistic
CRM	Coefficient of residual mass
d	Willmott's index of agreement
D	Kolmogorov-Smirnov's statistic
E	Mean relative error
EF	Modelling efficiency
EF1	Modified modelling efficiency
Fa <sub>2</sub>	Factor of two
FB	Fractional bias
$\mathbf{E}_{f}$	Fractional gross error
LC	Lack of correlation
LCS	Lack of positive correlation weighted by
	the standard deviations
LOFIT	Lack of statistical fit
MAE	Mean absolute error
MaxE	Maximum error
MB	Mean bias
MBE	Mean bias error
MdAE	Median absolute error
MG	Geometric mean bias
MSE	Mean square error
NMSE	Normalized mean square error
NU	Non-unity slope
PI	Range-based pattern Index
PI-F	F-based pattern index
PIdoy	Range-based pattern index versus day of
	year
PI <sub>Tmin</sub>	Range-based pattern index versus mini-
	mum air temperature
P(t)	Student's t-test probability
r	Pearson's correlation coefficient
r <sup>2</sup>	Least-square regression coefficient of de-
	termination

REF	Relative modelling efficiency
RMA	Reduced major axis
RMdAE	Relative median absolute error
RMSE	Root mean square error
ROC	Receiver-operator characteristic curve
RRMSE	Relative root mean square error
RMSV	Root mean square variation
SB	Simulation bias
SDSD	Square differences of the standard devia-
	tion
U	Theil's inequality coefficient
$U_B$	Systematic error proportion of Theil's
	inequality coefficient
$U_S$	Variance proportion of Theil's inequality
	coefficient
$U_C$	Covariance proportion of Theil's inequality
	coefficient
VG	Geometric mean variance

# **1** Introduction

The mathematical modelling of natural processes has undergone a large development during the last decades and, due to the complexity of the processes involved, this development is expected to pursue for a long time. The development of quantitative models to support the description of natural and semi-natural systems and decision-making in natural resource management is indeed considered to be of high priority. This is because models have a multitude of uses for scientists, managers and policy-makers investigating and governing natural processes. A major strength of models is in exploring interactions and feedback (e.g. Wainwright and Mulligan, 2004), helping to identify uncertainties and areas were we lack knowledge. They are also important supportive tools for the communication of complex issues to stakeholders of a non-scientific background. It is therefore important to demonstrate that a model has been tested using the most appropriate methods in order to achieve credibility with users of the estimates the model makes.

A mathematical model is, by definition, an approximate reconstruction of actual phenomena and an integration of natural processes into mathematical formulae. For agricultural and ecological systems and resource use (climate, land, vegetation...), a multitude of different theories coexist, not only

amongst disciplines (plant physiology, hydrology, climatology...), but also within disciplines (Argent, 2004; Beven, 2007; Arnold et al., 2008). Ecological, soil, meteorological and hydrological conditions in the actual systems are indeed the product of multiple concurrent processes, where multiple factors interact at different scales, each examined by different disciplines (Parker et al., 2003). This produces an abundance of theories and alternative explanations and, consequently, alternative models. One of the challenges is the process of bringing data and models together. It is required that numerical models should be preceded by thorough evaluation before use in practical applications, because the approximations used for the synthesis of a model often lead to discrepancies and deviations of the model results from nature.

Model evaluation is an essential step in the modelling process because it indicates if the implementation of the calculations involved reproduces the conceptual model of the system to be simulated (model reliability) and the level of accuracy of the model in reproducing the actual system (model usefulness) (Huth and Holzworth, 2005). Model evaluation includes any action in which the quality of a mathematical model is established (e.g., Metselaar, 1999; Jakeman et al., 2006). The topic of model evaluation has long attracted considerable debate amongst members of the scientific community. Much debate has stressed over the meaning of terms such as "testing", "validation", "verification" and "calibration" as part of the process collectively referred to as "evaluation" (Prisley and Mortimer, 2004).

The procedures to perform the evaluation task are also not widely accepted (Cheng et al., 1991) and appear in several forms, depending on data availability, system characteristics and researchers' opinion (Hsu et al., 1999). Biophysical, process-based models (unlike statistical models) are made up of mixtures of rate equations, comprise approaches with different levels of empiricism, aim at simulating systems which show a non-linear behaviour and often require numerical rather than analytical solutions. Figure 1 identifies two types of models: a conceptual model and a computerized model. The conceptual model is composed of information (input data, parameters and equations) that describes the physical system or process of interest. The computer program includes technical issues and possible errors. In practice, the computer program is tested, rather than the mathematical model representing the system (Leffelaar et al., 2003). Subjects such



**Fig. 1** Phases of modelling and simulation and the role of validation and verification (after Schlesinger, 1979)

as system representation (model structure) and program verification play a role besides numerical evaluation (structural assessment and program testing, not discussed in this paper) in assessing model accuracy (Donatelli et al., 2002b). Shaeffer (1980) developed a methodological approach to evaluate models that consisted of six tasks: (a) model examination, (b) algorithm examination, (c) data evaluation, (d) sensitivity analysis, (e) validation studies, and (f) code comparison studies.

This paper focuses on task (e) of Shaeffer's methodology, where the term validation (Sage, 1987) is used for the process of comparing model outputs with measurements, although terminology is not standardized. A comprehensive list of publications regarding model validation was compiled by Hamilton (1991) but the connotation "valid" is rather controversial. Validation implies that the model is correct, whereas models (in the same way as hypotheses) can only be falsified rather than proven (Bair, 1994; Oreskes, 1998). This is why the meaning of the terms is not entirely consistent across fields with some eschewing the use of the term validation (e.g., Anderson and Bates, 2001), others noting the problems implied by the term while acknowledging it is widespread use (e.g., Oreskes and Belitz, 2001), and other distinguishing numerous kinds of validation including operational, conceptual, data, and even processes (e.g., Rykiel Jr., 1996). It is also acknowledged in this paper (and conveniently adopted) that the term validation is widely reported in the literature and generally used as a universal term to denote model assessment.

Definitions of validation in relation to computer software and modelling have changed little over the

Table 1	Alternative	definitions	of model	(software)	) validation
---------	-------------	-------------	----------	------------	--------------

Definition	Source
"It is a valid and sound model if it accomplishes what is	Forrester (1961)
expected of it	Milerom (1072)
in adequacy of the model as a minic of the system which it	Minirain (1972)
is intended to represent	$0 \pm 1 \pm 1 \pm 1 \pm 1 \pm 1 \pm 1 \pm 1 \pm 1 \pm 1 \pm $
Substantiation that a computerized model within its domain	Schlesinger (1979)
of applicability possesses a satisfactory range of accuracy	
consistent with the intended application of the model"	
Comparison of model results with numerical data	American Society for Testing and Material (1984)
independently derived from experience or observations of the	
environment"	
"Validation means building the right system"	U Keefe et al. $(1987)$
The validation of a model package refers to the overall	Versar Inc. (1988)
process of defining the range of circumstances or situations	
for which the package's behaviour and predictions are satisfactory"	
To determine that it [the software] performs its intended	Wallace and Fujii (1989)
functions correctly, to ensure that it performs no unintended	
functions, and to measure its quality and reliability"	
The determination of the correctness of a model with respect	National Acid Precipitation Assessment Program (1990)
to the user's needs and requirements"	
The process of testing a computer program and evaluating	Institute of Electrical and Electronics Engineers (1991)
the results to ensure compliance with specific requirements"	
"The process of determining the degree to which a model is	American Institute of Aeronautics and Astronautics
an accurate representation of the real world from the	(1998)
perspective of the intended uses of the model"	
"A process of adding strength to our belief in the	Marcus and Elias (1998)
predictiveness of a model by repeatedly showing that it is not	
blatantly wrong in specific applications"	
"Having a conclusion correctly derived from premises"	Sterman (2000)
"The process of assessing the prediction ability"	Li et al. (2003)
"Substantiation that a model within its domain of applicability	Refsgaard and Henriksen (2004)
possesses a satisfactory range of accuracy consistent with the	
intended application of the model"	
"Examining whether the system achieved the project's stated	Sojda (2004)
purpose related to helping the user(s) reach a decision(s)"	
"To gauge how well a model performs against observed field	Huth and Holzworth (2005)
and laboratory measurements"	
"A procedure consisting in comparing the model output with	Dias and Lopes (2006)
field or laboratory data to prove the model efficiency"	
"Substantiating that the behavior of the model "mimics" the	Aumann (2008)
behaviour of the system with sufficient accuracy so that it is	
impossible to distinguish the behaviors of both systems in the	
experimental frames"	
"A procedure consisting in verifying if the model is able to	Cardoso and Lopes (2008)
reproduce data, independently of those involved in its	
calibration"	
"The assessment of the performance of a model against an	Matthews et al. (2008)
independently collected dataset"	

years. A summary of definitions is reported in Table 1, which though not absolute, are becoming more definite over time. It is apparent that the most recent definitions tend to mirror the use of the concept in 1960s–1970s, whereas 1980s' definitions were more computer-oriented. The most recent definitions are by and large adopted by this paper. Whilst such relatively simple definitions of all the issues pertaining to

581

validation can have their problems, they allow model community to communicate adequately enough in order to leave the semantic debate behind. It is worth noting that the understanding of validation is context dependant.

The overall model validation covers different areas, but the essence of it consists in defining criteria that will be taken into consideration in the choice of an "acceptable" model, and then testing the model performance according to those criteria. To assess the agreement between model results and observed data points, graphical plots are often made and judged qualitatively. It is acknowledged (e.g., Kuhnert et al., 2005) that if model output can be presented in appropriate formats, direct visual comparisons of models with data can yield significant insights about model performance. Statistical analysis by indices and test statistics play an important role to make comparisons reproducible, objective and quantitative. In general, however, the methodological basis for the validation of models to find the most suitable for specific applications is rudimentary, due to a lack of standardized terminology and procedures. Whilst statistical tools are easily applied for testing some empirical models, they might be of limited use with mechanistic (processbased) models whose primary aim is to represent system behaviour based on underlying principles (Berk et al., 2002). As modelling studies become more complex, models are used by parties less familiar with their peculiarities (e.g. structure, assumptions, data requirements and operability), and who may also lack understanding of the complexity of the entity being represented by the model. Some authors (e.g. Robinson and Ek, 2000) take the view that validation is the responsibility of the model user but improved accessibility to models by independent assessors and the ease with which they can be linked may increase their improper use. Hence, validation must not be seen as a oneoff event or a "once-and-for-all" activity (Janssen and Heuberger, 1995), but as an on-going process to check for model compatibility to current evidence and variations (e.g. in spatial, climatic and hydrological conditions). Moreover, according to Sinclair and Seligman (2000), demonstration that model output more or less fits a set of data is a necessary but not sufficient indication of validity because model validity is rather the capability to analyzing, clarifying, and solving empirical and conceptual problems. Empirical problems in a domain are, in general, about the observable world

in need of explanation because not adequately solved by a model, solved in different ways by rival models, or solved/unsolved depending on the model. Conceptual problems arise when the concepts within a model appear to be logically inconsistent, vague and unclear, or circularly defined, and when the definition of some phenomenon in a model is hard to harmonize with an ordinary language or definition (e.g. Parker, 2001). This raises the issue of widening beyond numerical testing by also including stakeholder evaluation and expert interpretation through soft systems approaches (Matthews et al., 2008). Working all this out would extend much further than the scope of this paper that is principally meant to recognise the limitations of numerical testing in achieving salience, legitimacy and credibility of models. Issues are initially discussed on the difficulties encountered when performing validation tests. Secondly, a review is given on how models are currently evaluated using qualitative and quantitative statistical techniques. Details are then provided on recent developments in validation criteria, decomposition of statistics and on how to combine validation statistics into single indicators of model performance. Principles and rules of general application are set forth with cited examples from the literature on ecology, agronomy, soil and climate modelling.

## 2 Issues on Model Validation

Before discussing specific approaches to validation, it is important to recognize a number of questions that arise in trying to validate a model. The following is a discussion of the main issues and factors as summarized in Table 2.

#### 2.1 Validation Purpose

There are many purposes for validation, including establishment of overall credibility in the model, assessment of how "right" or "wrong" a model is in a given application, along with production of evidence to support that a specific configuration of input data, parameter sets and model structure are appropriate for a particular application.

	Major factors to investigate					
Key validation issues	Modelling objective	Model inputs	Model outputs	Model structure	Modelling conditions	
Validation purpose	Х		Х		Х	
Robustness of results			Х		Х	
Interpretation of phenomena		Х	Х	Х		
Model comparison				Х		
Model predictions	Х		Х		Х	
Model complexity		Х	Х	Х		
Data accuracy		Х	Х			
Time histories			Х			

Table 2 Key validation issues and relative modelling features

A model may be appropriate in one context and not in another. The validation tests employed, therefore, also have to reflect the differing contexts of model application. Similarly, there is a need for model developers to understand the things that would make a model valuable to a model user. Hence, validation purpose is somewhat related to the purpose for which the model was created and used. Feedback from validation should provide valuable information to both the developers on how their model may be improved, but also to end users who need to know how confident they can be in the quality of outputs. The type of validation to be executed depends on the objective for which the model was developed or purpose to which the output is to be used. It cannot be assumed that a model that is valid for one purpose is also valid for another (Sargent, 2001). This dependence on purpose may explain why common criteria, standard terminology and formalized protocols are missing, and why subjective judgement is included in the validation process (Hamilton, 1991; Landry and Oral, 1993; Rykiel Jr., 1996). In crop modelling, it emerged out of the work of various scientific research teams that early focus was on providing a simulation capability for scientists to use in distinct agricultural situations. In recent years, however, the models have been increasingly used for informing policy development and even for real-time information support for land-managers (e.g. Hochman et al., 2005). This change in model application has led to a change in the focus in model testing. Huth and Holzworth (2005) appeal to how a growing user-base for a model (including users making real-time decisions) can place a greater importance on the need for testing for model reliability (ability of the calculations involved to reproduce the conceptual model) than for model usefulness (ability to reflect the behaviour of actual systems).

## 2.2 Interpretation of Phenomena

Different interpretations of the real world (Checkland, 1981) may present problems for validation. It is essential that model variables have the same definition as the actual data meant to be represented by the model itself. In simulating the developmental response of photosensitive plants, for instance, the ability to compute day-length is essential, but day-length can be defined in different ways depending upon the angle of the sun with the horizon (Forsythe et al., 1995). In modelling leaf area expansion it is not always clear if both simulation representation and data collection target the expanded part of a leaf (i.e. lamina expansion and growth) only, or account for the stem-like structure of a leaf that is attached to the stem (i.e. base and petiole expansion and growth) as well (e.g. Cornelissen et al., 2003). Measurements of plant development present a series of challenges as differences in assessing development can be due to the subjectivity of an observer or to a definition that is not unambiguously applied in the field (Hanft and Wych, 1982). Similarly, methods of determining kernel number, kernel mass, and yield can vary among researchers, which can add errors to comparisons between experimental results and simulated values (e.g. Anonymous, 2003). Such examples emphasise the importance of meta-data associated with original observations and development of model parameters (Medlyn and Jarvis, 1999).

Process-based models are also moving targets: if, for instance, plant model version 1 is considered to be incorrect, even a small change in a sub-model introduced to correct its functionality may produce a different interpretation on simulated processes (similar to the problem of "regression" in software development jargon). The reason for these unwanted changes lies in the lack of independence/wrong dependencies of parts of code, which is not completely avoidable. This aspect might go beyond a simple evaluation by once again comparing against previously acceptable results (Huth and Holzworth, 2005) and poses the need for formal model validation against observed data at each published stage of model development (Van Oijen, 2002).

#### 2.3 Model Comparison

Model comparison can be useful as a complement to model validation (Meehl et al., 2005). When two or more models are constructed for the same system or purpose, it is possible to make comparisons between them in order to select which is the best.

The Global Climate and Terrestrial Ecosystems group (GCTE) recognized in 1992 (GCTE, 1992) that there were at least 14 models of physiological processes that govern wheat growth and development. Landau et al. (1998, 1999, 2000) and Jamieson et al. (1999) reported on validation of three of such models against observed grain yields in the United Kingdom.

Diekkrüger et al. (1995) illustrated the simulation results from using simple as well as complex models against a common dataset covering the processes of water, nitrogen, plant growth and pesticide dynamics. In general, the models reproduced the measured dynamic only in part, with different response for different models. The study also made it clear that the experience of a scientist applying a model is as important as the difference between various model approaches.

When either field or reference modelled data are not available, attempts can be made to determine the proximity of one model to the other, also known as covalidation (Wright, 2001). Co-validation requires the assessment of the difference between models with respect to the values of their common output. The most likely case is that competing models do not share the same form. Typically, dissimilarly structured models not only have different inputs (including both variable and parameter sets), but they also have different levels of aggregation and capabilities in terms of modelling the actual system. In absence of actual data, the fact that several models show the same behaviour does not really give more confidence in each of them but only demonstrates that the models are capable of reproducing similar results for the observed system. This concept is often referred as equifinality (Beven, 1993; Franks et al., 1997; Beven and Freer, 2001; Medlyn et al., 2005). In general, potential "extra" capabilities of one model compared to another should not be used in co-validation. For example, nitrogen stress effects to plant growth should not be part of the comparison between two models where only one model includes nitrogen processes.

When statistical regression models (e.g. generalized linear models, generalized additive models) are compared, artificial data based on explicit theory can be used as "truth" (Austin et al., 2006).

## 2.4 Model Predictions

In model fitting, the model under investigation is evaluated for its adequacy in describing the observed data (Myung and Pitt, 2003). This is achieved by determining values for parameters in the model that will best describe the observations. Model fitting in this manner yields valuable estimates of quantities which might be used to differentiate between or explain process/system behaviours. However, papers on modelling often state that they aim to produce an instrument for prediction (Van Oijen, 2002). The issue of model prediction has been accompanied by some debate on the terminology to be used to describe model validation (see definitions by Marcus and Elias, 1998 and Li et al., 2003 in Table 1). In this case, a fundamental issue is to quantify the degree to which a model captures an underlying reality and predicts future cases.

According to Van Oijen (2002), model-based predictions only contribute to science if the underlying model mechanisms are described, are innovative, are compared to other approaches, and if the predictions can be checked and used as a test of the model. Predictions pose special problems for testing, especially if prediction focuses on events in the far future. Predictive models can be accepted if they explain past events (ex-post validation). However, the probability of making reasonable projections decreases with the length of time looked forward. A continuous exchange of validation data among developers and test teams should either ensure a progressive validation of the models by time, or highlight the need for updated interpretations of the changed system.

The problem of global change has generated much interest in the development of predictive models for crops and ecosystems (e.g. Intergovernmental Panel on Climate Change, http://www.ipcc.ch) because model estimates are increasingly being used for decision support and strategic planning (e.g. Matthews et al., 1999; Rivington et al., 2007; Tingem et al., 2009). This requires that the quality of model estimates is assessed in advance, or that the decision support outcomes be made insensitive to the estimation uncertainty (Norton, 2003). Model quality may also show variability over geographical locations. For example, Moberg and Jones (2004), in testing hindcast estimates produced by the Hadley Centre Regional Climate Model (http://www.metoffice.com/research/ hadleycentre) at 185 sites in Europe, found only some sites well represented. Responses like this restrict the geographical location to which the predicted climate change data could be used in model-based impact studies. Daily forecasts in hydrology largely use the previous week's real-time monitoring data as a training hindcast period, which permits validation and then project ahead for a period of two weeks (Quinn, 2008). In plant breeding, the issue of model-based prediction is dealt with when models are used as decision support tools to predict yield components of new cultivars and/or over new environments (Barbottin et al., 2006).

Like models which extrapolate in time, models used to extrapolate from small (e.g. leaf photosynthesis) to large spatial scales, such as regions, continents, or the global biosphere, are difficult to evaluate (e.g. Bolte et al., 2004; Chen and Coughenour, 2004). The heterogeneities in distributions of processes and non-linearity in their functional responses may make it infeasible to apply models representing these small scale processes over large areas and vice versa. There are several steps in performing spatial predictions where the variography, a well known geostatistical tool to analyse and to model anisotropic spatial correlations, is proposed to be used in the assessment of modelling results in addition to the traditional statistical analysis to demonstrate presence/absence of spatial structures in datasets (Kanevski et al., 2008).

## 2.5 Model Complexity

Process-based models reflect the same complexity as the system being represented (Bolte et al., 2004) and there are authors (e.g. Pilgram et al., 2002) who emphasize the need for models to accurately reflect the dynamics of the actual system. The comparison of model results with measured data is usually done at the system level and is thus the result of both the feedbacks within and between each of the sub-models making up the whole. Since possible counter-interaction between model components may go unnoticed, ideal validation should take place both at the level of sub-models and of the complete model (Leffelaar, 1990). As modelling projects progress, the model grows and it becomes harder to validate every part of it in detail. However, the independent testing of sub-units increases the possibility of building a complex system model as the assembly of interacting processes, rather than degrading a process-based model into a fitting exercise.

Systems with small timescale, spatial extension and complexity can be isolated, thus models of these systems are more readily accessible to validation. In complex models, specific sub-processes cannot always be tested at the level of the model output (Klepper, 1989). For systems with increasing time and spatial scales with increasing complexity, validation of system models requires increasingly more effort (Fig. 2). This increase in effort can be illustrated if one considers, for instance, validation requirements in terms of data, manpower, and organizational continuity for a model of an annual crop, for a model describing a crop rotation experiment, and for a model to analyze sustainable agricultural practices. At some point the effort required to gather validation data may no longer be feasible (De Wit, 1993).

The scientific literature is full of instances where the behaviour of complex systems is reduced to a set of basic emergent behaviours. The intrinsic complexity typically captured by many biophysical models may suggest a shift in emphasis from rigorous validation to a more exploratory approach characterizing the likelihood of distinct system behaviours rather than estimation of specific outputs (Bolte et al., 2004). The benefit of more flexible approaches to validation is clearly demonstrated in their ability to highlight model behaviour in extreme circumstances that might not appear in model validation datasets, but that certainly exist in the real world in which the model will be applied. This is the case, for instance, of high-magnitude events such as weather extremes that permit a form of model validation against past events, also called a "mental model" (Favis-Mortlock et al., 2001) based on the observer memory of landscape changes, i.e. massive soil degradation (with, possibly, the support of



· Harder to calibrate between sub-model components

Fig. 2 Trade-off between model complexity and validation

photographic material and annotated texts). Sensibility tests (Huth and Holzworth, 2005) refer to as the comparison of model output against more subjective, local experts feeling for what the model should do. Model responses to various stimuli are evaluated against the regularly observed system responses that are often captured in the notion of "common sense". Examples from local agronomists include, for instance, statements like "under those conditions the model should have an average yield of x t  $ha^{-1}$  with a range of y to z". These tests are a way of making sure the model performs in situations where observed data for validation are not available, and their use in testing the correctness of the underlying model design should not be undervalued. The "hard" data available for model testing is indeed only a small snapshot of system behaviour, and some authors (Seibert and McDonnell, 2002 for a catchment's model; Pastres et al., 2004 for sea grass model) showed that a better process representation can be obtained by using "soft" data. This might protect against being the model "right for the wrong reasons" due to adapting a model to a small set of observations. One might become "less right" but for the right reasons.

## 2.6 Data Accuracy and Quality

The accuracy of a model is determined on one hand by the authenticity of the algorithms describing the processes of the real world, while on the other hand by the quality of both its input data and data used to evaluate its outputs. Inaccuracies are common in both inputs and measured outputs. Model validation must therefore be accompanied by critical examination of the source and nature of the data used. Random errors may occur when single, individual samples do not take into account temporal variability; when samples taken at different points do not represent the actual area of interest; or when the inputs have been modified by unnoticed factors (environmental or human). Random errors of sample collection and processing may also occur when the sample is contaminated by other environmental media, when the sample is modified during transit and storage, or when the sample data are misrecorded. Systematic errors might occur if instruments are miscalibrated, measurements are taken at inappropriate locations or seasons, no measurements or estimates are made of relevant factors, or the data are not representative of the same output as the modelled one.

Complex models must generally be tested against a broad range of data of varying types, quality, and coverage. In reality, modellers are often confronted with a poor database because data monitoring is normally limited to a few points where samples are collected and analysed at some intermittent frequency. This may also change with the variables analysed, e.g. only scattered field measurements for some variables, more complete time series data or even maps for other variables, only qualitative assessments for others again. In hydrology, for instance, consistent field data of extreme events are poor (Westrich, 2008). Due to their sparse resolution, the data are frequently subject to large errors and this is en ever-increasing problem in moving from plots or small catchments to watersheds and regions.

Weather and soil variables are key inputs for biophysical models. Common problems encountered with weather inputs are: records taken too far from the experimental site, errors under reporting values, poor maintenance and imperfect sensitivity of sensors. Soil properties are often essential input data, but soil name and classification are commonly reported in the literature for field sites based solely on location and defined soil series from published data, rather than from field measurements. The soil may only marginally meet the classification criteria and therefore have different characteristics (pH, particle size distribution, organic matter content, etc.) than a soil meeting the central trend of the stated soil, illustrating the need for an appropriate sampling strategy to capture the spatial variability (Wright et al., 2003). Inaccuracies in both weather and soil inputs may turn into combined effects, either able to strengthen or weaken each other's effect during model simulation (Fodor and Kovács, 2003) and being a source of errors in model estimates (Hoogenboom, 2000; Heinmann et al., 2002; Rivington et al., 2003, 2006). This makes it difficult to disentangle the impacts on outputs of the combined effects of error source manifesting themselves as compensating errors. Emphasis must then be to determine the uncertainties that imperfect data introduce to model estimates.

As regards model outputs, comparison of model results with observations is complicated by the fact that biophysical data are affected by errors. It is difficult to appropriately evaluate model performance if the uncertainty in the data is high. Fitting a set of data can therefore only establish legitimacy for a model under the specific circumstances for which it was evaluated (Oreskes et al., 1994). The data resource is often segregated into two groups: a sample is used to estimate model parameters (model calibration: the adjustment of model parameters to improve the fit); an independent sample is used to assess the fit of the model to the data. Validating the model against data not used to construct the model is probably the best and simplest way to assess uncertainty and derive the reliability of model estimates (e.g. Ljung, 1999; Gardner and Urban, 2003). However, biophysical data often exhibit a large degree of variability, and it is common that there are important discrepancies between model estimates and actual data despite significant calibration efforts (e.g. Gobas et al., 1998). Dedicated techniques do exist to generate reduced bodies of data (e.g. Stone, 1974; Efron, 1986; Breiman and Spector, 1992). Cross-validation is a random data split into a number (commonly in between 5 and 10) of roughly equalsized parts; in turn, each part is used as a test set and the other parts for fitting the model. The prediction error is computed each time in the test set, and estimate of prediction error is the average of individual prediction errors. This technique is inappropriate for small sample size because the variance of the error is likely to increase considerably when splitting the data. The bootstrap technique is a valuable alternative (Wallach and Goffinet, 1989). A bootstrap sample is a sample created from random drawings with replacement from the original dataset (there can be repeats in a sample, and samples that do not contain an original point). It is also common in data-rich situations and for model comparison to generate three datasets: one to fit models, second part for estimating prediction error, third part for assessing prediction error of final selected model. As pointed out by Sinclair and Seligman (2000), a test of validity by separating data into sets may reflect only the success in splitting the data so that each group represents the same population of data. The same authors remarked: "a test of model performance is better served scientifically when based on all available data covering a wide range of conditions".

Extension of data series is also critical. Pastres et al. (2004) found that whilst traditional testing of their seagrass model gave an adequate description of the available model calibration data, the model failed to capture the known trends in sea grass evolution over a longer time frame.

Measurement uncertainty has important implications in modelling applications. Analysis of uncertainty in measured data which drive model calibration and validation improves model application and enhances decisions based on modelling results. However, the impact of uncertainty in model calibration and validation data is discussed but rarely included in the assessment of model accuracy. In order to change this omission, several goodness-of-fit indicators were modified to incorporate measurement uncertainty into model calibration and validation (Harmel et al., 2007, 2008).

#### 2.7 Robustness of Model Results

Model robustness is its reliability under different sets of experimental conditions. Lack of robustness in model results may reflect an absence of explicit reference to physical processes in the construction of mathematical relationships. Concern is voiced about the fact that many model assessments are frequently limited to a small number of output variables and sets of conditions (Woodward, 2001). Well-known validation methods, whilst effective for some things, are not necessarily effective at ensuring that a model faithfully captures the underlying process driving the system in question and thus will be applicable in any new situations where a model may be employed. Simple comparisons can be misleading, for example, when only the final "end of growing season" results from a crop model are compared, because reasonable estimations of overall growth can be achieved by different pathways, not all of which are logically acceptable (Sinclair and Seligman, 2000). Performance of a model should be checked not only in terms of the major outcome of the model, but also in the estimations for critical constituent components throughout the simulation. Such checks give protection against spurious conclusions and indicate whether the model is conceptually consistent and related to reality, or specific areas require improvement. The validation of a sugarcane model by Keating et al. (1999) as cited by Sinclair and Seligman (2000), gives an example of a sound test of robustness for the following reasons: (1) a large number (i.e. 19) of data sets for sugarcane growth were used, (2) a broad range of conditions with differing locations, irrigation treatments, and nitrogen fertility treatments were explored, (3) the seasonal evolution of individual components of the model (leaf area index, green biomass, millable stalk biomass, stalk sucrose, and nitrogen accumulation) was compared with observed results. Qualified features were the discussion of the situations where relatively large deviations existed between simulated and observed results, and the presentation of hypotheses to stimulate further research to improve the understanding contained in the current model. Such model validation serves as the basis for building credibility with end users, and greatly improves the probability that the model will be accepted.

#### 2.8 Time Histories

When simulating energy transfer or mass transformation in dynamic models, a time delay/anticipation frequently occurs if estimated versus measured values are compared. Peak synchronization between estimates and measurements most often will not occur. If synchronous comparison between estimates and measurements is applied, models which produce no response with respect to a specific process can yield better results, compared to models which show a time mismatch in the response (e.g. Vichi et al., 2004). In cases where poor parameterization produces a time shift of estimates, large residuals from few points may lead one to discount the model. Model re-parameterization can help correct time-related bias. If not, apparent misalignment of observed and modelled output may lead to a re-assessment of the daily time interval (common in many systems models) as the basis for comparing modelled and monitored event data (e.g. Yagow, 1997). Average values during multi-day periods (ten-day, month or year) can be used to calculate and compare occurrence of the event. This is particularly important where timing of events (i.e. crop phenology and management synchronization) estimated by a model are used to construct recommendations or optimized practises.

# 2.9 Summary

In this section, we summarized the vast international experience in the validation of biophysical models and ordered issues that we regard key to model validation into a list that may add value to the modelling work. Basically, we stressed that validation is purposedependent, based on equivalent definition of modelled and observed phenomena, to be substantiated over a variety of conditions (robustness), and possibly run at the level of individual processes in complex models. The discussion about data quality raises the need of a system for grading the relative quality of the input and the relative importance of the variables to be fit. Concerns were addressed regarding specific aspects such as predictions in the far future and synchronization of modelled and observed peak values. Moreover, model comparison was discussed as complementary to proper validation. Put into a logical structure, the ideas we have discussed are virtually applicable to validation of any model, and could equally be consistent with modelling in a variety of fields.

## 3 Validation of Models

A range of statistical measures and visual techniques can be used to assess goodness-of-fit of a given model and to compare the performance of a suite of models, as informed by the specific context of the problem (e.g., Berk et al., 2001). Recent review papers include: Mayer and Butler (1993), Janssen and Heuberger (1995), Smith et al. (1997), Martorana and Bellocchi (1999), Yang et al. (2000), Bellocchi (2004), Tedeschi (2006), Wallach (2006). Such papers provide and discuss a range of statistical measures and visual techniques that can be used to assess goodness-of-fit of a given model and to compare the performance of a suite of models. In this section we do not replicate a detailed examination of validation techniques, rather we consider validation approaches as developed by many authors and applied in numerous modelling studies in the domain of natural sciences. It is also beyond the scope of this paper to critically appraise in detail the usefulness of each method, as details are available in the review papers detailed above. The commonly used forms of numerical and statistical approaches are highlighted here with examples of criteria for determining model acceptability. The above cited papers (and the other papers in this section dealing with specific statistics) contain the equations of the statistics and methods used for model validation. So the equations are not reported here, but the rationale behind the choice of particular sets of statistics is given.

#### 3.1 Validation Measures

There are two main categories of goodness-of-fit measures for testing of one-at-a-time output, which are: (a) residual-based; and (b) association-based. Residualbased measures such as the mean bias error (MBE), and root mean square error (RMSE) provide quantitative estimates of the deviation of modelled outcomes from measurements. On the other hand, measures of statistical association such as the Pearson's correlation coefficient (r) provide quantitative estimates of the statistical co-variation between observed and estimated values (Addiscott and Whitmore, 1987). Statistical measures aim to characterize the usefulness of a model for a specific application and may lead users to decide whether or not to use a model for that particular application. Moreover, visual comparison of modelled and measured data, and experience-based judgement on the part of the modeller have been deemed important by researchers for assessing model validity and applicability in decision making.

Mean residuals: Mean bias (MB), the mean difference between observed and model-estimated values, is likely to be the oldest statistic to assess model accuracy (Cochran and Cox, 1957). More common is the mean square error (MSE), or equivalently its square root, the root mean square error (RMSE, or derived statistics such as the relative root mean square error RRMSE). MSE is also the statistic whose value is usually minimized during the parameter calibration process (Soroshian et al., 1993; Makowski et al., 2006). Mean absolute error (MAE) measures the mean absolute difference between observed and estimated values (Mayer and Butler, 1993), and is also used as the mean absolute percent error. Fox (1981) proposed to use: (1) RMSE or MAE to quantify the average difference between estimates and measurements; (2) the mean bias error (MBE) to identify under- or over-estimates; and (3) the variance of the distribution of the differences to quantify error variability. Such proposals are reflected in the paper of Davies and McKay (1989) and Trnka et al. (2005) for validation of solar radiation estimates.

**Modelling efficiency:** Willmott (1981) developed an index of agreement (d) to be used in addition to the previous measures. The modelling efficiency statistic (EF, Nash and Sutcliffe, 1970), interpreted as the proportion of variation explained by the model, has been extensively used in plant and hydrology models (e.g. Greenwood et al., 1985; Loague and Green, 1991), and can certainly be used in biological and ecological models. Mayer and Butler (1993), likewise, indicated both RMSE and the MAE as stable statistics, and recognized modelling efficiency (EF) as an important overall measure of fit. Smith et al. (1997) pointed out that EF and a coefficient of determination (CD) should be

used together for a better interpretation of RMSE when standard error of the measurements is unavailable. Alternative forms of the efficiency measures are given in Krause et al. (2005).

**Correlation:** The Pearson's correlation coefficient is largely used in validation. Fox (1981) and Willmott (1982) provided strong arguments against the use of this coefficient alone as a measure of performance. Its magnitude is indeed not consistently related to the accuracy of estimates, as correlation between dissimilar estimates and measurements can be high while, conversely, small differences between estimates and measurements may occur with low correlation values. Nonparametric correlation measures are also used for model validation such as concordance, Spearman and Kendall's coefficients (Press et al., 1992; Dhanoa et al., 1999; Agresti, 2002).

**Linear regression:** A linear regression between estimated and observed values is also commonly used. The hypothesis is that the regression passes through the origin and has a slope of unity (see Subsect. 3.3). The use of the  $r^2$  regression statistic (least-squares coefficient of determination) for model performance is flawed, as it does not account for model bias (Mayer and Butler, 1993; Mitchell, 1997). Krause et al. (2005) proposed to use  $r^2$  as a weighing factor of regression slope to quantify under- or over-estimates. An alternative (nonparametric) method to compute  $r^2$  was proposed by Kvalseth (1985), resulting in a coefficient that is more resistant to outliers or extreme data points.

**Combined graphical and statistical approaches:** The factor of two (Fa<sub>2</sub>) is currently used to evaluate air dispersion models (e.g. Kumar, 2000); combined with the values of different indices, MBE, fractional bias (FB), normalized mean square error (NMSE), correlation coefficient (r), geometric mean bias (MG), and geometric mean variance (VG). FB in the form of absolute differences is presented as fractional gross error ( $E_f$ ) in Seigneur et al. (2000).

Loague and Green (1991) suggested the use of both statistical and graphical measures for validation. Model performance can be compared using either summary statistics (mean, range, standard deviation) or using individual measured versus estimated pairs of data, which can also be displayed in both statistical and graphical forms. Assessment of data pairs usually proceeds with an analysis of the residual errors in the forms of maximum error (MaxE), RMSE, EF, CD and coefficient of residual mass (CRM) (James and Burges, 1982; Green and Stephenson, 1986). Suggested graphical displays include: (1) comparison of measurements and estimates; (2) comparison of ranges, medians and means; (3) comparison of matched estimated and measured time-series values and/or residuals; (4) comparison of cumulative values; and (5) cumulative frequency distributions.

Zacharias et al. (1996) presented robust quantitative techniques, from median-based nonparametric statistical methods (MdAE: median absolute error; RMdAE: relative median absolute error; REF: relative modelling efficiency), that can be used when the distribution of the observed data is non-Gaussian or when the sample size is not large enough to determine the underlying data distribution. Zacharias and Coakley (1993) categorized validation techniques into three main categories: summary statistics, hypothesis testing, and measures of goodness-of-fit, i.e. MaxE, a relative measure of the root mean square error (RRMSE), CD, EF and CRM. They listed examples of summary statistics as the mean, standard deviation, and those statistics commonly used with box or whisker plots (range, inter quartile range, and median).

Yang et al. (2000) examined the correlation across different statistics, allowing one to choose from each correlated group without loosing accuracy. They argued that the same conclusion can be achieved by using together either RMSE, modified modelling efficiency (EF1), paired t-test and E, or MAE, EF and E.

Model assessment by Mankin et al. (1977) and improved by Scholten and van der Tol (1998) was based on a comparison between model estimates and observations by using Venn diagrams and measures such as model adequacy (number of agreements between model and experiments / number of experiments) and reliability (number of agreements between model and experiments / number of model responses). These are helpful in discriminating between a better and worse model, and to define cases of useless or good models. Gardner and Urban (2003) illustrated a general approach to test model performance across a spectrum of methods via receiver-operator characteristic curves (ROC). Such approaches are based on the classification of results into discrete categories and the concept of a "confusion matrix" (Campbell, 1996), and imply defining false and true positives in the estimation of binary variables. Some statistics based on the confusion

matrix are presented and discussed in Beguería (2006). Pontius Jr. and Schneider (2001) described how to use the ROC as a quantitative measurement to validate a land-cover change model. In Barbottin et al. (2008), a ROC curve analysis was carried out to estimate the frequencies of correct and incorrect indicator-based and model-based decisions, using the area under the ROC curve as summary of the overall accuracy of a model.

**Patterns:** Change of patterns in the residuals can be assessed by testing the autocorrelation in the residuals (Vincent, 1998). Lin et al. (2002) developed modelchecking techniques by taking the cumulative sums of residuals over certain coordinates to ascertain whether or not specific patterns exist in the residual plot. Donatelli et al. (2000, 2004a) proposed to quantify the presence of patterns of residuals versus independent variables (e.g., a model input or a variable not considered in the model), by computing pattern indices of two types: range-based (PI) and F-based (PI-F). Macropatterns were revealed in model residuals by dividing the range of values of the external variable in two to five, fixed or varying sub-ranges. A pattern index in a percent relative-to-mean form was used as a validation measure by Bellocchi et al. (2003). In Trnka et al. (2006), estimated and observed herbage productions from permanent grassland were compared by using a vast array of pattern indices (against nitrogen fertilizer application rate, year, cut number, location, length of the growing season, date of the previous cut, number of snow days, two variants of accumulated air temperature, two variants of accumulated global solar radiation, and total precipitation during the period of sward growth) in conjunction with a set of performance statistics, i.e. MBE, RMSE, d, and Theil's inequality coefficient (U, Theil et al., 1970). The latter (ranging from 0 – perfect fit – to 1 – absence of any fit) penalizes large errors more than small ones and it also assesses a model's ability to duplicate a turning point or rapid changes in the data (Topp and Doyle, 2004). Pattern indices (versus month of year and minimum air temperature) in conjunction with error and correlation measures were also used by Diodato and Bellocchi (2007a) to assess the relative performance of three models of rainfall erosivity.

**Correction factors:** Correction factors were developed by Harmel and Smith (2007) for the error term in some goodness-of-fit indicators (modelling efficiency, index of agreement, root mean square error, and mean absolute error) to incorporate the uncertainty of measured data into model validation, later improved by Harmel et al. (2008) to consider the effect of model uncertainty.

In model predictions, one approach to estimating the prediction measures is to adjust the naïve measures to get less biased estimates of the prediction measures. Estimators that are based on this approach include the Mallows' statistic ( $C_p$ , Mallows, 1973), the Akaike's information criterion (AIC, Akaike, 1974), and the Bayesian information criterion (BIC, Schwartz, 1978). All of these estimators, functions of the naïve measures, size of the dataset, and number of parameters in the model, generally lack of robustness (Li et al., 2003).

#### 3.2 Disaggregating Statistics

Once parameters of linear regression of model estimates versus actual data are estimated, the fitted line can be applied to generate new estimates of the variable under study. The difference between model-based and regression-based estimates defines the erratic portion of the error (and is the basis for computation of prediction error), while the systematic portion is described by the difference between regression-based estimates and actual data (the basis to assess the precision of the fitted linear regression, Tedeschi, 2006). Both model users and developers will focus on reducing the systematic error, the formers by model re-calibration, and the latters by better defining the basic equations. This is a basic concept by Aitken (1973) and Willmott (1981). More recently, Kobayashi and Salam (2000) developed the same concept to have residuals disaggregated into erratic and systematic components. They used the root mean square variation (RMSV) to indicate how much the model fails to estimate the variability of the measures around the mean, together with derived measures such as simulation bias (SB), square differences of the standard deviations (SDSD) and lack of positive correlation weighted by the standard deviations (LCS). These statistics are supportive in locating the causes of possible large deviations between estimates and measurements. The proportional contribution of systematic and erratic portions to the total error is helpful in determining areas in the model requiring further improvement. Further developing those findings, a different partitioning of mean square error into three additive components was given by Gauch and Fick (2003) and Gauch et al. (2003): they retained SB and derived the non-unity slope (NU), and the lack of correlation (LC). There is a unique quality to these approaches in the way the authors commented them in a letter exchange (Gauch et al., 2004; Kobayashi, 2004).

Trnka et al. (2006) called attention to disaggregating the Theil's coefficient (U). One of the main advantages of coefficient U is the possibility of calculating proportions of: estimated bias (U<sub>B</sub>), indicating systematic errors; variance (U<sub>S</sub>) that measures the ability of the model to replicate the degree of variability in the data; covariance (U<sub>C</sub>), that is any remaining error, after accounting for the bias and variance effects. The ideal distribution of inequality over these three sources is for the bias and variance effects equal to zero, and the covariance equal to one.

#### 3.3 Statistical Hypothesis Tests

Hypothesis testing is a formal approach to validation where either summary statistics or goodness-of-fit measures are tested against prescribed criteria (range of accuracy). In using statistical hypothesis testing to assess the validity of a model for its intended application, two hypotheses are formulated under the given set of experimental conditions: for the null hypothesis model is valid for an acceptable range of accuracy, and for the alternative hypothesis it is invalid under the same acceptable range of accuracy. Accepting the alternative hypothesis when the null hypothesis is true corresponds to the type-I statistical error, whose probability is also called model builder's risk. Accepting the null hypothesis when the alternative hypothesis is true matches the type-I statistical error, the second type of wrong decision with a probability called model user's risk (Balci and Sargent, 1982a).

**Fundamental requirements:** Statistical tests assume that the outcomes based on the model are statistically accurate representations of what they purport to estimate. However, both systematic and random errors in biophysical studies may influence the accuracy of the estimators (as seen in Subsect. 2.7). Pennell et al. (1990) stated that graphical analyses allow for

identification of trends in the data, systematic errors, and other potential sources of error, such as outliers. Marcus and Elias (1998) elaborated on five major areas of concern when applying formal statistical tests: observational data may not have been collected for the purposes of model validation; the sample size may be too small, allowing inadequate power to detect model deficiencies or to discriminate among competing models; the sample size may be so large that even useful models may be rejected by a statistical test for deviations that have little practical importance; measurement errors may bias the test statistics in the direction of attenuating the apparent goodness of the model; the temporal rhythms of the output variable may be influenced by systematic and random errors occurring in environmental factors.

Typical forms of the goodness-of-fit test are the following: Does the observed value minus modelestimated value equal zero (showing that the estimates are unbiased)? Does the ratio of observed value to model-estimated value equal one (in which case the estimates are relatively unbiased)? Numerous approaches involving statistical analysis have been used to evaluate model adequacy and several forms of statistical hypotheses are structured to show the level of confidence that the hypothesis is not rejected.

Using regressions: The regression between observed and model-estimated values is commonly used because estimates of the intercept and the slope are good indicators of accuracy (the simultaneously closer to zero and unity, respectively, the higher the accuracy). Nonetheless, necessary assumptions have to be considered when performing a linear regression: the X-axis values are known without error; the Y-axis values have to be independent, random and with equal variance; residuals are independent and identically distributed. Some authors (e.g. Bland and Altman, 1995; Kleijnen et al., 1998) critically revisited the role of regression analysis in model validation, and suggested alternative approaches (difference between simulated and actual outputs against their sum, against their average, etc.) for achieving non-misleading regression testing. Fila et al. (2003a, b) suggested using the reduced major axis (RMA) method in place of the ordinary least squares method to estimate regression parameters. RMA has three desirable properties (Ricker, 1984): it is symmetric in X and Y (if the x and y axes are interchanged, the slope is replaced by its reciprocal and the line remains

stationary about the data points); it is scale independent (the line does not change with a change of scale); it is robust to clusters of observations in the frequency distributions of data (the line usually describes the central trend even when the sample is not bivariate normal).

Thomann (1982) suggested using regression analysis and tests of slope (against one) and intercept (against zero) of a regression line in conjunction with other performance statistics. If the model is a good one, the regression will be a 45° line thorough the origin. Thus, the adequacy of the model can be determined by testing if intercept equals zero and slope equals one, separately using Student t-tests, or simultaneously using the F-test (Dent and Blackie, 1979; Mayer et al., 1994). These parametric tests for whether the regression line is significantly different from the 1:1 line assume that the data are normally distributed and independent, which is often not the case for data determined by non-linear biophysical processes.

**Interpretation of statistics:** The test statistics may also have ambiguous interpretations. A statistical test of observed versus estimated values which fails to achieve the desired level of confidence does not necessarily indicate a problem in the model (usually the specification of one or more key parameters), because the problem may (or may also) reside in the observed data (usually the result of measurement error). With the t-tests for the intercept and slope, the more scatter in the data points, the greater is the standard error of the regression parameters, the smaller is the computed value for the test statistic and therefore, the harder it is to reject the null hypotheses which states the parameters are equal to zero and one respectively. Therefore, the test can fail to reject the null hypothesis either because the regression parameters are really not different from the values desired or there is much scatter around the line (Harrison, 1990). The F-based calculation that tests if the intercept and the slope coefficients simultaneously are respectively not different from zero and unity is affected by the same ambiguity as the t-tests. As shown by Analla (1998), the greater the estimation error the more difficult it is to reject the null hypothesis. Alternatively, the confidence interval should be used to investigate the range of the slope (Mitchell, 1997).

Empirical confidence intervals were proposed by Parrish and Smith (1990) as practical test for model validity founded on an overlap between the ranges of values computed on both model outputs and observations. Upper and lower limits of the range were computed as division and multiplication of the nominal model estimate by a chosen factor. The factor of two (Fa<sub>2</sub>) is commonly used in air dispersion modelling (Kumar, 2000). Summaries of confidence limits on normalized mean square error, geometric mean variance and geometric mean bias were used by Patel and Kumar (1998) to select the best out of three air dispersion models.

When replicated experiments are available, the lack-of-statistical-fit (LOFIT) can be calculated and tested against an F-distribution (Whitmore, 1991). Assuming random experimental errors, the LOFIT distinguishes the mean deviation as a source of error from the failure of the model. Smith et al. (1997) assessed the statistical significance of difference-based indices assuming a deviation corresponding to a given confidence interval of the measurements. Fila et al. (2003a, b) proposed statistical methods to compare experiments against estimates when both are replicated. The use of bootstrapping techniques for validation of simulation models when parametric assumptions are violated was proposed by Kleijnen et al. (2001).

Non-parametric tests: Contrary to parametric analyses, with nonparametric tests (i.e. variants of the Kolmogorov-Smirnov test as described in Stephens, 1974; the Wilcoxon-signed rank test, as described in Daniel, 1995) the assessment of adequacy of a model is related to its ability to yield the same ranking between observed and model-estimated values rather than model-estimates on observed values per se. Reckhow and Chapra (1983) listed measures of error, the t-test, the non-parametric Wilcoxon test, regression analysis, cross-correlation and box plots as appropriate statistical methods for deterministic models. Reckhow et al. (1990) also recommended various combinations of graphic (bivariate plots, histograms, and box plots) and statistical procedures based on the proposed analysis and intended use of modelling results. The  $\chi^2$  tests described by Agresti (2002) indicate whether the data points are homogenously distributed or if there is any tendency of over- or under-estimation.

The comparison of the distribution of the observed and model-estimated values has also been utilized to identify model adequacy for stochastic (Reynolds and Deaton, 1982) and deterministic models (Dent and Blackie, 1979). The common Kolmogorov-Smirnov's D test has been used to assess the probability that two data sets (observed and model-estimated values) have the same distribution. It consists to measure the overall difference of the area between two cumulative distribution functions (Press et al., 1992).

The use of multivariate statistics in model validation is not new (Balci and Sargent, 1982b). With multidimensional models, multivariate statistical analyses can be proficiently applied as in Mabille and Abecassis (2003), where a geometric model of wheat grain morphology was evaluated via principal component analysis and discriminating factorial analysis, and generating confidence limits in an elliptical plane.

## 3.4 Validation Criteria

One of the difficulties when evaluating assessment metrics is determining what values indicate "good" or "bad" models. This section outlines the development of criteria (not statistical) that have been applied within a range of published studies. Clouse and Heatwole (1996) stated that "primary usefulness is in assessing which modelling scenarios are better predicted than other scenarios". However, other authors have taken a different approach by setting definitive criteria for several statistics.

Criteria used in James and Burges (1982) included CD and EF > 0.97 for good hydrological model performance. Dillaha (1990) stated that good hydrologic model assessment should estimate observed values within a factor of two, where parameters are measured on site, or where the model is calibrated and within a factor of 10 otherwise. Kumar (2000) used a criterion that air dispersion model estimates be within a factor of two, by looking at the percentage of estimates meeting such a criterion (Fa<sub>2</sub>  $\geq$  80%), combined with: NMSE  $\leq$  0.5, -0.5  $\leq$  FB  $\leq$  +0.5, 0.75  $\leq$  MG  $\leq$  1.25, and 0.75  $\leq$  VG  $\leq$  1.25.

The general categorization for the range of values of Pearson's correlation coefficient (r) by Hinkle et al. (1994) may indicate a straightforward (nonstatistically based) way of interpreting the calculated correlation between estimates and measurements: 0.0 to 0.3, little (very weak) if any correlation; 0.3 to 0.5, low (weak) correlation; 0.5 to 0.7, moderate correlation; 0.7 to 0.9, high (strong) correlation; 0.9 to 1.0 very high (very strong) correlation. In the Erosion Productivity Impact Calculator validation performed by Chung et al. (1999), the following criteria were chosen to assess if the model results were satisfactory: RMSE and MdAE < 50%, EF and REF> 0.3, -0.2 < CRM +0.2. Standards of <20% for the percentage error and >0.5 for r<sup>2</sup> were also set. In a following paper (Chung et al., 2000) the target criteria to judge if the model results were satisfactory were: EF >  $0.3, r^2 > 0.5$ , and *P*-value of the paired t-test between the observed and simulated values >0.025.

In Stöckle et al. (1999) RRMSE and Willmott's d index were taken together to evaluate the weather generator ClimGen. Upper and lower limits were suggested for a solid judgment on the model performance: good when RRMSE  $\leq 10\%$  and d  $\geq 0.95$ , acceptable when  $10\% \leq \text{RRMSE} \leq 20\%$  and  $d \geq 0.90$ , poor with other values. In a following paper, Stöckle et al. (2004) adopted similar standards (but in a more restrictive fashion) for evaluating evapotranspiration estimates:  $d \ge 0.95$  and RRMSE  $\le 10\%$ , very good;  $d \ge 0.95$  and  $15\% \ge \text{RRMSE} > 10\%$ , good;  $d \ge 0.95$  and  $20\% \ge$ RRMSE > 15%, acceptable;  $d \ge 0.95$  and  $25\% \ge$ RRMSE > 20%, marginal. Other combinations of d and RRMSE values indicated poor performance. In addition, all combinations with slope >1.1 or <0.9 and  $r^2 < 0.85$  of the regression observed versus estimated values (forced through zero) were considered poor.

## 3.5 Combining Multiple Statistics

It emerges from the discussion above that each individual performance measure will only assess a single component of model behaviour and is not sufficient to judge model adequacy to a given purpose. It is possible that a model can be deemed unsuitable and rejected based on an assessment by one statistic assessing one form of model performance, whilst other attributes of the model may be desirable, as assessed by other measures. Similarly, a model may be seen as acceptable based on one performance statistic but still contains poor qualities not assessed by appropriate tests. The use of multiple metrics allows a greater range of model estimate behaviour to be tested, but still leaves the issue of how to achieve an overall model validation, i.e. weighing up the balance of positive and negative attributes.

The combination of multiple assessment metrics and the setting of criteria have evolved into formal structures, becoming attractive and being regarded as a positive step in achieving robust assessments. Bellocchi et al. (2002a) dealt with the need for integrated methods for model validation by introducing the concept of fuzzy multiple-metric assessment expert system to aggregate different metrics into one indicator for use in solar radiation model validation. The fuzzy system reflects the author's expert judgment about the quality of model performance. This approach enables the calculation of a single indicator made up of a number of individual metrics representing different indices or test statistics. Such an approach provides a comprehensive assessment, making it easier to identify best performing models.

The method exists as a flexible, open-ended structure (Fig. 3) in which a range of metrics can be aggregated into a single modular indicator, based on an expert weighting expression of the balance of importance of the individual indices and their aggregation into modules. The Bellocchi et al. (2002a) study used: relative root mean square error (RRMSE), modelling efficiency (EF), the probability of paired Student-t test (P(t)); the correlation coefficient of the estimates versus measurements (r) and the two pattern indices detailed in Donatelli et al. (2004a), one computed versus day of year ( $PI_{doy}$ ), and the other versus minimum temperature (PI<sub>Tmin</sub>). Values of each statistic are computed and then aggregated into modules: Accuracy (RRMSE, EF, P(t)); Pattern (PI<sub>dov</sub>, PI<sub>Tmin</sub>); and Correlation (r). A module is a validation measure calculated via a fuzzy-based procedure from one or more basic statistics. For each module, a dimensionless value between 0 (best model response) and 1 (worst model response) is calculated. The method adopts the Sugeno approach of fuzzy inference (Sugeno, 1985). Three membership classes can be defined for all indices, according to expert judgment, i.e. favourable, unfavourable and partial (or fuzzy) membership, using S-shaped curves as transition probabilities in the range favourable to unfavourable. A two-stage design of a fuzzy-based rules inferring system is applied, where firstly several metrics are aggregated into modules and then, using the same procedure, the modules are aggregated in a second level integrated index (again, ranging



**Fig. 3** Structure of the fuzzy-based integrated index for solar radiation model assessment (F = favourable; U = unfavourable; S = membership function; a = minimum value of F; b = maximum value of U; after Rivington et al., 2005)

from 0 to 1), called *indicator*. The expert reasoning runs as follows: if all input variables are favourable, the value of the module is 0 (good agreement between estimates and measurements); if all indices are unfavourable, the value of the module is 1 (poor agreement), while all the other combinations assume intermediate values. The weights can be chosen based on the users own experience in handling each statistic. In Bellocchi et al. (2002a) a decreasing importance was assigned to the modules: Accuracy, Pattern and Correlation. Rivington et al. (2005), Diodato and Bellocchi (2007b, c) and Abraha and Savage (2008) demonstrated the value of employing this method, in conjunction with using graphical illustrations, to gain a fine level of detail about model quality. Similarly to this, Donatelli et al. (2004b) developed integrated indices for evaluating the estimates from pedotransfer functions.

A fuzzy-based methodology was also proposed (Donatelli et al., 2002a) to identify mismatches between estimated and measured time series by using a fuzzy-based approach. The mismatch in time series comparison of LEACHM soil nitrogen estimates was identified by means of an integrated index derived aggregating RMSE and  $PI_{dov}$  (values  $\leq 2$  as favourable and  $\geq 12$  as unfavourable for both), 0.8 and 0.2 being the relative weights, and calculated reiterating the computation over a 100-day shift of model estimates. The same approach was integrated with a complementary set of statistics from the environmental modelling literature (e.g. Environmental Protection Agency, 1991) to evaluate SUNDIAL soil mineral nitrogen estimates (Bellocchi et al., 2004). Bellocchi et al. (2002b) extended the original fuzzy-based multiplemetric assessment system approach to aggregating the RRMSE values (values  $\leq 20\%$  as favourable and ≥40% as unfavourable) computed over different outputs in cropping systems modelling under different sets of conditions, thus allowing a comprehensive assessment of the model's performance by means of one integrated index. They attributed major weight to above ground biomass (i.e., 2), where a minor incidence of soil variables (water content: 1, nitrate content: 0.5) was recognized.

Aggregating measures of performance have in common that the information contained in the errors is aggregated into a single numerical value. Herbst and Casper (2008) argued that essentially different model results can be obtained with close to identical performance measure values. Because of their low discriminatory power, performance measures might not be well suitable to give evidence of the difference or equivalence between alternative model realizations. As a step towards improved extraction of information from existing data they introduced an approach, the Self-Organizing Map (SOM), which uses a number of performance statistics and represents, for each of them, the colour-coded spectrum of model realizations obtained from Monte-Carlo simulations. It was applied to a distributed conceptual watershed model. SOM is a type of artificial neural network and unsupervised learning algorithm that is used for clustering, visualization and abstraction of multi-dimensional data. Such an algorithmic approach mainly targets the optimization and identification of parameters that mostly affect the model output (sensitivity analysis), and is not of direct interest for this review.

## 3.6 Summary

Because of the vast collection and diversity of the approaches to assess models, we sorted through and straightened out validation statistics showing how they were introduced and applied to biophysical modelling. The review of the measures of performance that are commonly used in model validation reveals the positions assumed over time in the modelling literature mostly emphasising that a single statistic will only reveal one aspect of model performance. As each approach has its advantages and drawbacks, they are rather complementary and are generally used in combination in model validation. What values for assessment metrics indicate satisfactory models remains a subjective issue and no definitive guidance exists because of heterogeneity of approaches and application domains. While agreeing with many authors that model validation has to be performed using a set of dissimilar validation statistics, we in particular advocate the use of combined multiple statistics where several measures for validation can be considered both separately (each individual metric) and collectively (integrated indicator). Test statistics may be problematic because they rely on assumptions that are difficult to check in biophysical systems. Decomposition of statistics in

basic terms may disclose the characteristic and the actual structure of the error, but the combination of multiple metrics into synthetic indicators where subjective choices (expert decisions) are converted into explicit and transparent rules reveals a more comprehensive picture. The lack of precise and undisputable criteria to consider a specific metric as more effective than others, and the multiplicity of aspects to be accounted for a multi-perspective evaluation of model performance, logically leads to some use of composite metrics for model validation. A composite metric is not the only output of composition: the modeller can "drill down" to module values, and finally to basic metrics to better understand the synthetic result provided by the composite indicator. In such respect, composition of metrics should be considered a shift of paradigm from merely selecting the best out of a set of evaluation metrics.

## **4** Conclusion

This paper discusses issues concerned with model validation and reviews the most commonly used forms of estimate testing. Exposition of material and explanation of concepts presented in Section 2 ("Issues on model validation") reflect the authors' perception of issues that are fundamental to understanding the factors that are related to model validation. The examples provided throughout the text demonstrate how previous instances of model use (and success or failure associated with that use) are the growing knowledge bases acquired from using different models for various applications. The publications cited show how the scope and capabilities of validation approaches have evolved and improved with time. Though finding solution of how best to evaluate numerical values produced by models will remain an issue, a range of approaches do exist for improving the testing of model estimates. Our hope is that these approaches will continue to evolve.

Our historical reconstruction of the approaches serving the validation purposes, as presented in Section 3 ("Validation of models"), points towards three main outcomes achieved: disaggregation of validation statistics into basic components, introduction of validation criteria, and combination of statistics into synthetic indicators. Baseline thresholds of validation measures (extracted from the international literature and recapped in sub-section "Validation criteria") provide users with the modellers' perception of good/bad performance statistics. Such criteria are presented and discussed not only to make available reference values of possible use in future validation studies, but also they call on the need for using expert rules to guide the validation process. This review of the methods available for numerical testing has shown that greater value can be gained through combined use and rule-based aggregation of multiple approaches to achieve a more complete form of validation.

Advancements in these numerical testing methodologies for validation need to be put into structured frameworks comprised of processes such as sensitivity and uncertainty analyses (parameter and input variable appraisal), parameter optimization, model structure assessment (expert review), software testing, etc. As such, validation must be seen as an integral part of the overall model development and application process, whilst also encompassing the requirement for better data quality control and meta data recording. This places a greater emphasis on the need to include validation plans within model project proposals and a higher level of support for validation work by funding organisations. As shown, models may come in a variety of time and space resolutions and scales. Matching these scales and ensuring consistency in the overall model is not a trivial process and may be difficult to fully automate. Techniques to validate models need to be developed at the same pace with which the models themselves are created, improved and applied. Also, validation steps must be clearly stated, accessible, transparent, and understandable to non-modellers. As discussed in the context of the current knowledge, this can be achieved by means of reliability statistics, history of previous use, or personal preferences. However, details about validation techniques development go beyond the aim of this review, and a second paper on this broad topic may be arranged later as a natural evolution of what has already been presented.

Acknowledgements This paper was developed with the support of the Italian Ministry of Agricultural Policies; project SI-PEAA (http://www.sipeaa.it), paper No. 36. The authors also gratefully acknowledge the support of the Scottish Government Environment Rural Affairs Department. Thanks also to David Elston and David Walker of Biomathematics and Statistics Scotland (Macaulay Institute, Aberdeen, UK) for constructive criticism and suggestions.

#### References

- Abraha M.G., Savage M.J. (2008) Comparison of estimates of daily solar radiation from air temperature range for application in crop simulations, Agr. Forest Meteorol. 148, 401–416.
- Addiscott T.M., Whitmore A.P. (1987) Computer simulation of changes in soil mineral nitrogen and crop nitrogen during autumn, winter and spring, J. Agr. Sci. 109, 141–157.
- Agresti A. (2002) Categorical data analysis, (2nd ed.), Wiley, New York, NY, USA.
- Aitken A.P. (1973) Assessing systematic errors in rainfall runoff models, J. Hydrol. 20, 121–136.
- Akaike H. (1974) A new look at the statistical model identification, IEEE T. Automat. Contr. 19, 716–723.
- American Institute of Aeronautics and Astronautics (1998) Guide for the verification and validation of computational fluid dynamics, American Institute of Aeronautics and Astronautics, AIAA-G-077-1998, Reston, VA, USA.
- American Society for Testing and Material (1984) Standard practice for evaluating environmental face models of chemicals, American Society for Testing and Material, Philadelphia, PA, USA, Standard E 978–984.
- Analla M. (1998) Model validation through the linear regression fit to actual versus predicted values, Agr. Syst. 57, 115–119.
- Anderson M.G., Bates P.D. (2001) Hydrological science: model credibility and scientific understanding, in: Anderson M.G., Bates P.D. (Eds.), Model validation: perspectives in hydrological science, John Wiley & Sons, New York, USA, Vol. 1, pp. 1–10.
- Anonymous (2003) How to estimate grain harvest losses, Prairie Grains, Issue 54.
- Argent R.M. (2004) An overview of model integration for environmental applications – components, frameworks and semantics, Environ. Modell. Softw. 19, 219–234.
- Arnold T., Berger T., Uribe T. (2008) Step by step calibration of an integrated system for irrigation management, in: Quinn N.W.T. (Ed.) Integration of sensor networks and decision support tools for basin-scale, real-time water quality management, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 584–591.
- Aumann C.A. (2008) A methodology for building credible models for policy evaluation, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 1025–1032.
- Austin M.P., Belbin L., Meyers J.A., Doherty M.D., Luoto M. (2006) Evaluation of statistical models used for predicting plant species distributions: role of artificial data and theory, Ecol. Model. 199, 197–216.
- Bair E.S. (1994) Model (in)validation a view from courtroom, Ground Water 32, 530–531.
- Balci O., Sargent R.G. (1982a) Some examples of simulation model validation using hypothesis testing, in: Highland H.J., Chao Y.W., Madrigal O. (Eds.), Proc. 14th Conference on

Winter Simulation, December 6–8, San Diego, CA, USA, Vol. 2, pp. 621–629.

- Balci O., Sargent R.G. (1982b) Validation of multi-variate response simulation models by using Hotelling's two-sample T<sup>2</sup> test, Simulation 39, 185–192.
- Barbottin A., Le Bail M., Jeuffroy M.H. (2006) The Azodyn crop model as a decision support tool for choosing cultivars, Agron. Sustain. Dev. 26, 107–115.
- Barbottin A., Makowski D., Le Bail M., Jeuffroy M.-H., Bouchard C., Barrier C. (2008) Comparison of models and indicators for categorizing soft wheat fields according to their grain protein contents, Eur. J. Agron. 29, 175–183.
- Beguería S. (2006) Validation and evaluation of predictive models in hazard assessment and risk management, Nat. Hazards 37, 315–329.
- Bellocchi G. (2004) Appendix A. Numerical indices and test statistics for model evaluation, in: Pachepsky Ya., Rawls W.J. (Eds.), Development of pedotransfer functions in soil hydrology, Elsevier, Amsterdam, The Netherlands, pp. 394–400.
- Bellocchi G., Acutis M., Fila G., Donatelli M. (2002a) An indicator of solar radiation model performance based on a fuzzy expert system, Agron. J. 94, 1222–1233.
- Bellocchi G., Donatelli M., Fila G. (2003) Calculating reference evapotranspiration and crop biomass using estimated radiation inputs, Ital. J. Agron. 7, 95–102.
- Bellocchi G., Fila G., Donatelli M. (2002b) Integrated evaluation of cropping systems models by fuzzy-based procedure, in: Villalobos F.J., Testi L. (Eds.), Proc. 7th European Society for Agronomy Congress, 15–18 July, Cordoba, Spain, pp. 243–244.
- Bellocchi G., Smith J., Donatelli M., Smith P. (2004) Improvements in time mismatch analysis of model estimates, in: Jacobsen S.E., Jensen C.R. Porter J.R. (Eds.), Proc. of 8th European Society for Agronomy Congress, 11–15 July, Copenhagen, Denmark, pp. 221–222.
- Berk R.A., Bickel P., Campbell K. (2002) Workshop on statistical approaches for the evaluation of complex computer models, Stat. Sci. 17, 173–192.
- Berk R.A., Fovell R.G., Schoenberg F., Weiss R.E. (2001) The use of statistical tools for evaluating computer simulations – an editorial essay, Climatic Change 51, 119–130.
- Beven K.J. (1993) Prophecy, reality and uncertainty in distributed hydrological modelling, Adv. Water Resour. 16, 41–51.
- Beven K.J. (2007) Towards integrated environmental models of everywhere: uncertainty, data and modelling as a learning process, Hydrol. Earth Syst. Sc. 11, 460–467.
- Beven K.J., Freer J. (2001) Equifinality, data assimilation, and uncertainty estimation in mechanistic modelling of complex environmental systems using the GLUE methodology, J. Hydrol. 249, 11–29.
- Bland J.M., Altman D.G. (1995) Comparing methods of measurement: why plotting against standard method is misleading, Lancet 346, 1085–1087.
- Bolte J.P., Hulse D.W., Gregory S.V., Smith C. (2004) Modelling biocomplexity - actors, landscapes and alternative futures, in: Pahl-Woslt C., Schmidt S., Rizzoli A.E., Jakeman A.J. (Eds.), Complexity and integrated resources, Trans. 2nd Biennial Meeting of the International Environmental Modelling and Software Society, 14–17 June, Osnabrück, Germany, Vol. 1, pp. 1–10.

- Breiman L., Spector P. (1992) Submodel selection and evaluation in regression: the X-random case, Int. Stat. Rev. 60, 291–319.
- Campbell J.B. (1996) Introduction to remote sensing, 2nd ed., The Guilford Press, New York, NY, USA.
- Cardoso A.C., Lopes J.F. (2008) 3D ecological modelling of the Aveiro coast (Portugal), in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 181–190.
- Checkland P.B. (1981) Systems thinking, systems practice, John Wiley & Sons, London.
- Chen D.X., Coughenour M.B. (2004) Photosynthesis, transpiration, and primary productivity: Scaling up from leaves to canopies and regions using process models and remotely sensed data, Global Biogeochem. Cy. 18, GB4033.
- Cheng R.T., Burau J.R., Gartner J.W. (1991) Interfacing data analysis and numerical modelling for tidal hydrodynamic phenomena, in: Parker B.B. (Ed.), Tidal hydrodynamics, John Wiley & Sons, New York, NY, USA, pp. 201–219.
- Chung S.W., Gasman P.W., Huggins D.R., Randall G.W. (2000) Evaluation of EPIC for three Minnesota cropping systems. Working paper 00-WP 240, Centre for Agricultural and Rural Development, Iowa State University, Ames, IO, USA.
- Chung S.W., Gasman P.W., Kramer L.A., Williams J.R., Gu R. (1999) Validation of EPIC for two watersheds in Southwest Iowa, J. Environ. Qual. 28, 971–979.
- Clouse R.W., Heatwole C.D. (1996) Evaluation of GLEAMS considering parameter uncertainty, ASAE paper No. 96-2023, St. Joseph, MI, USA.
- Cochran W.G., Cox G.M. (1957) Experimental design, John Wiley & Sons, New York, NY, USA.
- Cornelissen J.H.C., Lavorel S., Garnier E., Diaz S., Buchmann N., Gurwich D.E., Reich P.B., ter Steege H., Morgan H.D., van der Heijden M.G.A., Pausas J.G., Poorter H. (2003) A handbook of protocols for standardised and easy measurement of plant functional traits worldwide, Aust. J. Bot. 51, 335–380.
- Daniel W.W. (1995) Biostatistics: a foundation for analysis in the health sciences, John Wiley & Sons Inc., New York, NY, USA.
- Davies J.A., McKay D.C. (1989) Evaluation of selected models for estimating solar radiation on horizontal surfaces, Sol. Energy 43, 153–168.
- De Wit C.T. (1993) Philosophy and terminology, in: Leffelaar P.A. (Ed.), On systems analysis and simulation of ecological processes – with examples in CSMP and FORTRAN. Kluver, Dordrecht, The Netherlands, pp. 3–9.
- Dent J.B., Blackie M.J. (1979) Systems simulation in agriculture, Applied Science Publishers Ltd., London, United Kingdom.
- Dhanoa M.S., Lister S.J., France J., Barnes R.L. (1999) Use of mean square prediction error analysis and reproducibility measures to study near infrared calibration equation performance, J. Near Infrared Spec. 7, 133–143.
- Dias, J.M., Lopes, J.F. (2006) Implementation and assessment of hydrodynamic, salt and heat transport models: the case of Ria de Aveiro Lagoon (Portugal), Environ. Modell. Softw. 21, 1–15.

- Diekkrüger B., Söndgerath D., Kersebaum K.C., McVoy C.V. (1995) Validity of agroecosystem models applied to the same data set, Ecol. Model. 81, 3–29.
- Dillaha T.A. (1990) Role of best management practices in restoring the health of the Chesapeake Bay: Assessments of effectiveness, in: Haire M., Krome E.C. (Eds.), Perspectives on the Chesapeake Bay, 1990: Advances in estuarine sciences. CBP/TRS41/90. Chesapeake Bay Consortium. US EPA Chesapeake Bay Program. Annapolis, Maryland, USA, pp. 57–81.
- Diodato N., Bellocchi G. (2007a) Estimating monthly (R)USLE climate input in a Mediterranean region using limited data, J. Hydrol. 345, 224–236.
- Diodato N., Bellocchi G. (2007b) Modelling reference evapotranspiration over complex terrains from minimum climatological data, Water Resour. Res. 43, doi:10.1029/ 2006WR005405.
- Diodato N., Bellocchi G. (2007c) Modelling solar radiation over complex terrains using monthly climatological data, Agr. Forest Meteorol. 144, 111–126.
- Donatelli M., Acutis M., Bellocchi G. (2000) Two statistical indices to quantify patterns of errors produced by models, in: Christen O., Ordon F. (Eds.), Proc. 3rd International Crop Science Conference, 17–22 August, Hamburg, Germany, p. 186.
- Donatelli M., Acutis M., Bellocchi G., Fila G. (2004a) New indices to quantify patterns of residuals produced by model estimates, Agron. J. 96, 631–645.
- Donatelli M., Acutis M., Fila G., Bellocchi G. (2002a) A method to quantify time mismatch of model estimates, in: Villalobos F.J., Testi L. (Eds.), Proc. 7th European Society for Agronomy Congress, 15–18 July, Cordoba, Spain, pp. 269–270.
- Donatelli M., Acutis M., Nemes A., Wösten H. (2004b) Integrated indices for pedotransfer function evaluation, in: Pachepsky Ya., Rawls W.J. (Eds.), Development of pedotransfer functions in soil hydrology. Elsevier, Amsterdam, The Netherlands, pp. 363–390.
- Donatelli M., van Ittersum M.K., Bindi M., Porter J.R. (2002b) Modelling cropping systems – highlights of the symposium and preface to the special issues, Eur J. Agron. 18, 1–11.
- Efron B. (1986) how biased is the apparent error rate of a prediction rule, J. Am. Stat. Assoc. 81, 461–470.
- Environmental Protection Agency (1991) Guideline for regulatory application of the urban airshed model. U.S., Environmental Protection Agency Office of Air Quality Planning and Standards, Research Triangle Park, NC, 27711, EPA-450/4-91-013.
- Favis-Mortlock D., Boardman J., MacMillan V. (2001) The limits of erosion modeling: why we should proceed with care, in: Harmon R.S., Doe W.W. III (Eds.), Landscape erosion and evolution modeling. Kluwer Academic/Plenum Publisher, New York, NY, USA, pp. 477–516.
- Fila G., Bellocchi G., Acutis M., Donatelli M. (2003a) IRENE: a software to evaluate model performance, Eur. J. Agron. 18, 369–372.
- Fila G., Bellocchi G., Donatelli M., Acutis M. (2003b) IRENE\_DLL: A class library for evaluating numerical estimates, Agron. J. 95, 1330–1333.
- Fodor N., Kovács G.J. (2003) Sensitivity of 4M model to the inaccuracy of weather and soil input data, Appl. Ecol. Environ. Res. 1, 75–85.

- Forrester J.W. (1961) Industrial dynamics, Pegasus Communications, Waltham, MA, USA.
- Forsythe W.C., Rykiel E.J. Jr., Stahl R.S., Wu H., Schoolfield R.M. (1995) A model comparison for daylength as a function of latitude and day of year, Ecol. Model. 80, 87–95.
- Fox D.G. (1981) Judging air quality model performance: a summary of the AMS workshop on dispersion models performance, Bull. Am. Meteorol. Soc. 62, 599–609.
- Franks SW., Beven K.J., Quinn P.F., Wright I.R. (1997) On the sensitivity of soil-vegetation-atmosphere transfer (SVAT) schemes: equifinality and the problem of robust calibration, Agr. Forest Meteorol. 86, 63–75.
- Gardner R.H., Urban D.L. (2003) Model validation and testing: past lessons, present concerns, future prospects, in: Canham C.D., Cole J.J., Lauenroth W.K. (Eds.), Models in ecosystem science. Princeton University Press, Princeton, NJ, USA, pp. 184–203.
- Gauch H.G. Jr., Fick W. (2003) In model evaluation, what is X and what is Y? in: 2003 annual meeting abstracts. [CD-ROM], ASA, CSSA and SSSA, Madison, WI, USA.
- Gauch H.G. Jr., Gene Hwang J.T., Fick G.W. (2003) Model evaluation by comparison of model-based predictions and measured values, Agron. J. 95, 1442–1446.
- Gauch H.G. Jr., Gene Hwang J.T., Fick G.W. (2004) Reply, Agron. J. 96, 1207–1208.
- Global Climate and Terrestrial Ecosystems (1992) Effects of change on the wheat ecosystem, Workshop report, GCTE Focus 3 meeting, Saskatoon, SK, Canada, 22–24 July, University of Oxford, United Kingdom.
- Gobas F.A.P.C., Pasternak J.P., Lien K., Duncan R.K. (1998) Development and field validation of a multimedia exposure assessment models for waste load allocation in aquatic ecosystems: application to 2,3,7,8-tetrachlorodibenzo-pdioxin and 2,3,7,8-tetrachlorodibenzofuran in the Fraser River watershed, Environ. Sci. Technol. 32, 2442–2449.
- Green I.R.A., Stephenson D. (1986) Criteria for comparison of single event models, J. Hydrol. Sci. 31, 395–411.
- Greenwood D.J., Neeteson J.J., Draycott A. (1985) Response of potatoes to N fertilizer: dynamic model, Plant Soil 85, 185–203.
- Hamilton M.A. (1991) Model validation: an annotated bibliography, Comm. Stat. Theor. M. 20, 2207–2266.
- Hanft J.M., Wych R.D. (1982) Visual indicators of physiological maturity of hard red spring wheat, Crop Sci. 22, 584–588.
- Harmel R.D., Smith P.K. (2007) Consideration of measurement uncertainty in the evaluation of goodness-of-fit in hydrologic and water quality modelling, J. Hydrol. 337, 326–336.
- Harmel R.D., Smith D.R., King K.W., Slade R.M., Smith P. (2008) Data uncertainty estimation tool for hydrology and water quality (DUET-H/WQ): estimating measurement uncertainty for monitoring and modelling applications, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 574–583.
- Harrison S.R. (1990) Regression of a model on real-system output: an invalid test of model validity, Agr. Syst. 34, 183–190.
- Heinmann A.B., Hoogenboom G., Chojnicki B. (2002) The impact of potential errors in rainfall observations on the simu-

lation of crop growth, development and yield, Ecol. Model. 157, 1–21.

- Herbst M., Casper M.C. (2008) Towards model evaluation using Self-Organizing Maps, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 3, pp. 1055–1062.
- Hinkle D., Wiersma W., Jurs S. (1994) Applied statistics for the behavioural sciences, 3rd ed., Houghton Mifflin Company, Boston, MT, USA.
- Hochman Z., van Rees H., Carberry P.S., Holzworth D., Dalgliesh, N.P., Hunt J., Poulton P.L., Brennan, L.E., Darbas T., Fisher J., van Rees, S., Huth N.I. Peake A.S., McCown R.L. (2005) Can access to a cropping system simulator help farmers reduce risk in drought-prone environments? in: InterDrought-II, 2nd Int. Conf. Integrated Approaches to Sustain and Improve Plant Production Under Drought Stress 24–28 September, Rome, Italy.
- Hoogenboom G. (2000) Contribution of agro-meteorology to the simulation of crop production and its applications, Agr. Forest Meteorol. 103, 137–157.
- Hsu M.H., Kuo A.Y., Kuo J.T., Liu W.C. (1999) Procedure to calibrate and verify numerical models of estuarine hydrodynamics, J. Hydrol. Eng. 125, 166–182.
- Huth N., Holzworth D. (2005) Common sense in model testing, in: Zerger A., Argent R.M. (Eds.), Proc. MODSIM 2005 International Congress on Modelling and Simulation: Advances and applications for management and decision making, 12–15 December, Melbourne, Australia, pp. 2804–2809.
- Institute of Electrical and Electronics Engineers (1991) IEEE standard glossary of software engineering terminology, IEEE, IEEE Std 610.12-1990, New York, NY, USA.
- Jakeman A.J., Letcher R.A., Norton J.P. (2006) Ten iterative steps in development and evaluation of environmental models, Environ. Modell. Softw. 21, 606–614.
- James L.D., Burges S.J. (1982) Selection, calibration, and testing of hydrologic models, in: Haan C.T., Johnson H.P., Brakensiek D.L. (Eds.), Hydrologic modelling of small watersheds, American Society of Agricultural Engineers, St. Joseph, MI, USA, pp. 437–472.
- Jamieson P.D., Porter J.R., Semenov M.A., Brooks R.J., Ewert F., Ritchie J.T. (1999) Comments on "Testing winter wheat simulation models predictions against observed UK grain yields" by Landau et al. (1998), Agr. Forest Meteorol. 96, 157–161.
- Janssen P.H.M., Heuberger P.S.C. (1995) Calibration of processoriented models, Ecol. Model. 83, 55–66.
- Kanevski M., Pozdnoukhov A., Timonin V. (2008) Machine learning algorithms for geospatial data. Applications and software tools, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 320–327.
- Keating B.A., Robertson M.J., Muchow R.C., Huth N.L. (1999) Modelling sugarcane production systems. 1. Development and performance of the sugarcane module, Field Crops Res. 61, 253–271.

- Kleijnen J.P.C., Bettonvil B., van Groenendaal W. (1998) Validation of trace-driven simulation models: a novel regression test, Manage. Sci. 44, 812–819.
- Kleijnen J.P.C., Cheng R.C.H., Bettonvil B. (2001) Validation of trace-driven simulation models: bootstrapped tests, Manage. Sci. 47, 1533–1538.
- Klepper O. (1989) A model of carbon flow in relation to macrobenthic food supply in the Oosterschelde estuary (S.W. Netherlands), PhD-Thesis, Wageningen Agricultural University, The Netherlands.
- Kobayashi K. (2004) Comments on another way of partitioning mean squared deviation proposed by Gauch et al. (2003), Agron. J. 96, 1206–1207.
- Kobayashi K., Salam M.U. (2000) Comparing simulated and measured values using mean squared deviation and its components, Agron. J. 92, 345–352.
- Krause P., Boyle D.P., Bäse F. (2005) Comparison of different efficiency criteria for hydrological model assessment, Adv. Geosci. 5, 89–97.
- Kuhnert M., Voinov A., Seppelt R. (2005) Comparing raster map comparison algorithms for spatial modeling and analysis, Photogramm. Eng. Rem. S. 71, 975–984.
- Kumar A. (2000) Dispersion and risk modelling, Department of Civil Engineering, University of Toledo, OH, USA, CIVE-6630:995.
- Kvalseth T.O. (1985) Cautionary note about R2, Am. Stat. 39, 279–285.
- Landau S., Mitchell R.A.C., Barnett V., Colls J.J., Craigon J., Moore K.L., Payne R.W. (1998) Testing winter wheat simulation models' predictions against observed UK grain yields, Agr. Forest Meteorol. 89, 85–99.
- Landau S., Mitchell R.A.C., Barnett V., Colls J.J., Craigon J., Payne R.W. (1999) Response to "Comments on 'Testing winter wheat simulation models predictions against observed UK grain yields by Landau et al. [Agr. For. Meteorol. 89 (1998) 85-99]' by Jamieson et al. [Agr. For. Meteorol., this issue]", Agr. Forest Meteorol. 96, 163–164.
- Landau S., Mitchell R.A.C., Barnett V., Colls J.J., Craigon J., Payne R.W. (2000) A parsimonious, multiple-regression model of wheat yield response to environment, Agr. Forest Meteorol. 101, 151–166.
- Landry M., Oral M. (1993) In search of a valid view of model validation for operations research, Eur. J. Oper. Res. 66, 161–167.
- Leffelaar P.A. (1990) On scale problems in modelling: an example from soil ecology, in: Rabbinge R., Goudriaan J., van Keulen H., Penning de Vries F.W.T., van Laar H.H. (Eds.), Theoretical production ecology: reflections and prospects. Simulation Monographs 34, Pudoc, Wageningen, The Netherlands, pp. 57–73.
- Leffelaar P.A., Meike H., Smith P., Wallach D. (2003) Modelling cropping systems – highlights of the symposium and preface to the special issues. 3. Session B. Model parameterisation and testing, Eur. J. Agron. 18, 189–191.
- Li W., Arena V.C, Sussman N.B.. Mazumdar S. (2003) Model validation software for classification models using repeated partitioning: MVREP, Comput. Meth. Prog. Bio. 72, 81–87.
- Lin D.Y., Wei L.J., Ying Z. (2002) Model-checking techniques based on cumulative residuals, Biometrics 58, 1–12.
- Ljung L. (1999) System identification Theory for the user, 2nd ed., Prentice Hall, Upper Saddle River, NJ, USA.

- Loague K., Green R.E. (1991) Statistical and graphical methods for evaluating solute transport models: overview and application, J. Contam. Hydrol. 7, 51–73.
- Mabille F., Abecassis J. (2003) Parametric modelling of wheat grain morphology: a new perspective, J. Cereal Sci. 37, 43–53.
- Makowski D., Hillier J., Wallach D., Andrieu B., Jeuffroy M.-H. (2006) Parameter estimation for crop models, in: Wallach D., Makowski D., Jones J.W. (Eds.), Working with dynamic models. Evaluation, analysis, parameterization and applications, Elsevier, Amsterdam, pp. 101–150.
- Mallows C. (1973) Some comments on Cp, Technometrics 15, 661–675.
- Mankin J.B., O'Neill R.V., Shugart H.H., Rust B.W. (1977) The importance of validation in ecosystem analysis, in: Innis G.S. (Ed.), New directions in the analysis of ecological systems, Proc. Series Simulation Council 5(1), La Jolla, CA, USA, pp. 63–72.
- Marcus A.H., Elias R.W. (1998) Some useful statistical methods for model validation, Environ. Health Persp. 106, 1541–1550.
- Martorana F., Bellocchi G. (1999) A review of methodologies to evaluate agro-ecosystems simulation models, Ital. J. Agron. 3, 19–39.
- Matthews K.B., Rivington M., Blackstock K., Buchan K., Miller D.G. (2008) Raising the bar – Is evaluating the outcomes of decision and information support tools a bridge too far? in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 948–955.
- Matthews K.B., Rivington M., Buchan K., Miller D., Bellocchi G. (2008) Characterising and communicating the agro-meteorological implications of climate change scenarios to land management stakeholders, Climate Res. 37, 59–75.
- Matthews K.B., Sibbald A.R., Craw S. (1999) Implementation of a spatial decision support system for rural land use planning: integrating GIS and environmental models with search and optimisation algorithms, Comput. Electron. Agr. 23, 9–26.
- Mayer D.G., Butler D.G. (1993) Statistical validation, Ecol. Model. 68, 21–32.
- Mayer D.G., Stuart M.A., Swain A.J. (1994) Regression of realworld data on model output: an appropriate overall test of validity, Agr. Syst. 45, 93–104.
- Medlyn B.E., Jarvis P.G. (1999) Design and use of a database of model parameters from elevated CO<sub>2</sub> experiments, Ecol. Model. 124, 69–83.
- Medlyn B.E., Robinson A.P., Clement R., McMurtrie E. (2005) On the validation of models of forest CO<sub>2</sub> exchange using eddy covariance data: some perils and pitfalls, Tree Physiol. 25, 839–857.
- Meehl G.A., Covey C., McAvaney B., Latif M., Stouffer R.J. (2005) Overview of the coupled model intercomparison project, Bull. Am. Meteorol. Soc. 86, 89–93.
- Metselaar K. (1999) Auditing predictive models: a case study in crop growth, PhD-Thesis, Wageningen Agricultural University, Wageningen.

- Mihram G.A. (1972) Some practical aspects of the verification and validation of simulation models, Oper. Res. Quart. 23, 17–29.
- Mitchell P.L. (1997) Misuse of regression for empirical validation of models, Agr. Syst. 54, 313–326.
- Moberg A., Jones P.D. (2004) Regional climate model simulations of daily maximum and minimum near-surface temperatures across Europe compared with observed station data 1961-90, Clim. Dynam. 23, 695–715.
- Myung J., Pitt M.A. (2003) Model fitting, in: Nadel L. (Ed.), The encyclopedia of cognitive science, Vol. 3, MacMillan, London, United Kingdom, pp. 47–51.
- Nash J.E., Sutcliffe J.V. (1970) River flow forecasting through conceptual models, Part I - A discussion of principles, J. Hydrol. 10, 282–290.
- National Acid Precipitation Assessment Program (1990) Evaluation of regional acidic deposition models and selected applications of RADM. Acid deposition: state of sciences and technology, The National Acid Precipitation Assessment Program, Washington, DC, USA, Vol. I, Report 5.
- Norton J.P. (2003) Prediction for decision-making under uncertainty, in: Post D.A. (Ed.), Proc. MODSIM 2003 International Congress on Modelling and Simulation: Integrative modelling of biophysical, social and economic systems for resource management solutions, 14–17 July, Townsville, Australia, Vol. 4, pp. 1517–1522.
- O'Keefe R.M., Balci O., Smith E.P. (1987) Validating expert system performance, IEEE Expert 2, pp. 81–90.
- Oreskes N. (1998) Evaluation (not validation) of quantitative models, Environ. Health Persp. 106, 1453–1460.
- Oreskes N., Belitz K. (2001) Philosophical issues in model assessment, in: Anderson M.G., Bates P.D. (Eds.), Model validation: perspectives in hydrological science, John Wiley & Sons, New York, NY, USA, Vol. 3, pp. 23–41.
- Oreskes N., Shrader-Frechette K., Belitz K. (1994) Verification, validation and confirmation of numerical models in the earth sciences, Science 263, 641–646.
- Parker V.T. (2001) Conceptual problems and scale limitations of defining ecological communities: a critique of the CI concept (Community of Individuals), Perspect. Plant Ecol. Evol. Syst. 4, 80–96.
- Parker D., Manson S., Janssen M., Hoffman M., Deadman P. (2003) Multi-agents systems for the simulation of land-use and land-cover change: a review, Ann. Assoc. Am. Geogr. 93, 314–337.
- Parrish R.S., Smith C.N. (1990) A method for testing whether model predictions fall within a prescribed factor of true values, with an application to pesticide leaching, Ecol. Model. 51, 59–72.
- Pastres R., Brigolin D., Petrizzo A., Zucchetta M. (2004) Testing the robustness of primary production models in shallow coastal areas: a case study, Ecol. Model. 179, 221–233.
- Patel V.C., Kumar A. (1998) Evaluation of three air dispersion models: ISCST2, ISCLT2, and Screen2 for mercury emissions in an urban area, Environ. Monit. Assess. 53, 259–277.
- Pennell K.D., Homsby A.O., Jessup R.E., Rao K.S.C. (1990) Evaluation of five simulation models for predicting aldicarb and bromide behaviour under field conditions, Water Resour. Res. 26, 2679–2693.

- Pilgram B., Judd K., Mees A. (2002) Modelling the dynamics of nonlinear time series using canonical variate analysis, Physica D 170, 103–117.
- Pontius R.G. Jr., Schneider L.C. (2001) Land-cover change model validation by an ROC method for the Ipswich watershed, Massachusetts, USA, Agr. Ecosyst. Environ. 85, 239–248.
- Press W.H., Teukolsky S.A., Vetterling W.T., Flannery B.P. (1992) Numerical recipes in Fortran 77: the art of scientific computing, 2nd ed., Cambridge University Press, New York, NY, USA.
- Prisley S.P., Mortimer M.J. (2004) A synthesis of literature on evaluation of models for policy applications, with implications for forest carbon accounting, Forest Ecol. Manag. 198, 89–103.
- Quinn N.W.T. (2008) Integration of sensor networks and decision support tools for basin-scale, real-time water quality management, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 44–53.
- Reckhow K.H., Chapra S.C. (1983) Engineering approaches for lake management. Vol. 1: Data analysis and empirical modelling, Butterworth Publishers, Boston.
- Reckhow K.H., Clements J.T., Dodd R.C. (1990) Statistical evaluation of mechanistic water-quality models, J. Environ. Eng. 116, 250–268.
- Refsgaard J.C., Henriksen H.J. (2004) Modelling guidelinesterminology and guiding principles, Adv. Water Resour. 27, 71–82.
- Reynolds J.M.R., Deaton M.L. (1982) Comparisons of some tests for validation of stochastic simulation models, Commun. Stat. Simul. Comput. 11, 769–799.
- Ricker W.E. (1984) Computation and uses of central trend lines, Can J. Zool. 62, 1897–1905.
- Rivington M., Bellocchi G., Matthews K.B., Buchan K. (2005) Evaluation of three model estimations of solar radiation at 24 UK stations, Agr. Forest Meteorol. 135, 228–243.
- Rivington M, Matthews K.B., Bellocchi G., Buchan K. (2006) Evaluating uncertainty introduced to process-based simulation model estimates by alternative sources of meteorological data, Agr. Syst. 88, 451–471.
- Rivington M, Matthews K.B., Bellocchi G., Buchan K., Stöckle C.O., Donatelli M. (2007) An integrated assessment approach to conduct analyses of climate change impacts on whole-farm systems, Environ. Modell. Softw. 22, 202–210.
- Rivington M., Matthews K.B., Buchan K. (2003) Quantifying the uncertainty in spatially-explicit land-use model predictions arising from the use of substituted climate data, in: Post D.A. (Ed.), Proc. MODSIM 2003 International Congress on Modelling and Simulation: Integrative modelling of biophysical, social and economic systems for resource management solutions, 14–17 July, Townsville, Australia, Vol. 4, pp. 1528–1533.
- Robinson A.P., Ek A.R. (2000) The consequences of hierarchy for modelling in forest ecosystems, Can. J. Forest Res. 30, 1837–1846.
- Rykiel Jr. E.J. (1996) Testing ecological models: the meaning of validation, Ecol. Model. 90, 229–244.

- Sage A.P. (1987) Validation, in: Singh M.G. (Ed.), Systems analysis & control encyclopaedia: theory, technology, applications, Pergamon, Oxford, United Kingdom.
- Sargent R.G. (2001) Verification, validation and accreditation of simulation models, in: Peters B.A., Smith J.S., Medeiros D.J., Rohrer M.W. (Eds.), Proc. 2001 Winter Simulation Conference, December 10–13, Arlington, VA, USA, pp. 106–114.
- Schlesinger S. (1979) Terminology for model credibility, Simulation 32, 103–104.
- Scholten H., van der Tol M.W.M. (1998) Quantitative validation of deterministic models: when is a model acceptable? in: Society for Computer Simulation (Ed.), Proceedings of the Summer Computer Simulation Conference, San Diego, CA, USA, pp. 404–409.
- Schwartz G. (1978) Estimating the dimension of a model, Ann. Stat. 6, 461–464.
- Seibert J., McDonnell J.J. (2002) On the dialog between experimentalist and modeler in catchment hydrology: use of soft data for multicriteria model calibration, Water Resour. Res. 38, 1241.
- Seigneur C., Pun B., Pai P., Louis J.F., Solomon P., Emery C., Morris R., Zahniser M., Worsnop D., Koutrakis P., White W., Tombach I. (2000) Guidance for the performance evaluation of three-dimensional air quality modeling systems for particulate matter and visibility, J. Air Waste Manage. Assoc. 50, 588–599.
- Shaeffer D.L. (1980) A model evaluation methodology applicable to environmental assessment models, Ecol. Model. 8, 275–295.
- Sinclair T.R., Seligman N. (2000) Criteria for publishing papers on crop. modelling, Field Crop. Res. 68, 165–172.
- Smith P., Smith J.U., Powlson D.S., McGill W.B., Arah J.R.M., Chertov O.G., Coleman K., Franko U., Frolking S., Jenkinson D.S., Jensen L.S., Kelly R.H., Klein-Gunnewiek H., Komarov A.S., Li C., Molina J.A.E., Mueller T., Parton W.J., Thomley J.H.M., Whitmore A.P. (1997) A comparison of the performance of nine soil organic matter models using datasets from seven long-term experiments, Geoderma 81, 153–225.
- Sojda R.S. (2004) Empirical evaluation of decision support systems: concepts and an example for trumpeter swan management, in: Pahl-Woslt C., Schmidt S., Rizzoli A.E., Jakeman A.J. (Eds.), Complexity and integrated resources, Trans. 2nd Biennial Meeting of the International Environmental Modelling and Software Society, 14–17 June, Osnabrück, Germany, Vol. 2, pp. 649–655.
- Soroshian S., Duan Q., Gupta V.K. (1993) Calibration of rainfall-runoff models: application of global optimization to Sacramento Soil Moisture Model, Water Resour. Res. 29, 1185–1194.
- Stephens M.A. (1974) EDF statistics for goodness of fit and some comparisons, J. Am. Stat. Assoc. 69, 730–737.
- Sterman J.D. (2000) Business dynamics: systems thinking and modeling for a complex world, Irwin McGraw-Hill, New York, NY, USA.
- Stöckle C.O., Bellocchi G., Nelson R.L. (1999) Evaluation of the weather generator ClimGen for several world locations, in: Bindi M., Donatelli M., Porter J., van Ittersum M.K. (Eds.),

Proc. 7th International Congress for Computer Technology in Agriculture, 15–18 November 1998, Florence, Italy, pp. 34–41.

- Stöckle C.O., Kjelgaard J., Bellocchi G. (2004) Evaluation of estimated weather data for calculating Penman-Monteith reference crop evapotranspiration, Irrig. Sci. 1, 39–46.
- Stone M. (1974) Cross-validatory choice and assessment of statistical predictions, J. R. Stat. Soc. Ser. B-Stat. Methodol. 36, 111–147.
- Sugeno M. (1985) An introductory survey of fuzzy control, Inf. Sci. 36, 59–83.
- Tedeschi L.O. (2006) Assessment of the adequacy of mathematical models, Agr. Syst. 89, 225–247.
- Theil H., Cramer J.S., Moerman H., Russchen A. (1970) Economic forecast and policy, 2nd ed., North-Holland Publishing Company, Amsterdam, The Netherlands.
- Thomann R.V. (1982) Verification of water quality models, J. Env. Eng. Div. 108, 923–940.
- Tingem M., Rivington M., Bellocchi G., Azam-Alia S., Colls J. (2009) Adaptation assessments for crop production in response to climate change in Cameroon, Agron. Sustain. Dev. 29, in press.
- Topp C.F.E., Doyle C.J. (2004) Modelling the comparative productivity and profitability of grass and legume systems of silage production in northern Europe, Grass Forage Sci. 59, 274–292.
- Trnka M., Eitzinger J., Gruszczynski G., Buchgraber K., Resch R., Schaumberger A. (2006) A simple statistical model for predicting herbage production from permanent grassland, Grass Forage Sci. 61, 253–271.
- Trnka M., Zãlud Z., Eitzinger J., Dubrovský M. (2005) Global solar radiation in Central European lowlands estimated by various empirical formulae, Agr. Forest Meteorol. 131, 54– 76.
- Van Oijen M. (2002) On the use of specific publication criteria for papers on process-based modelling in plant science, Field Crop. Res. 74, 197–205.
- Versar Inc. (1988) Current and suggested practices in the validation of exposure assessment models, Office of Health and Environmental Assessment, United States environmental Protection Agency, Washington DC, USA.
- Vichi M., Ruardij P., Baretta J.W. (2004) Link or sink: a modelling interpretation of the open Baltic biogeochemistry, Biogeosciences 1, 79–100.
- Vincent L.A. (1998) A technique for the identification of inhomogeneities in Canadian temperature series, J. Climate 11, 1094–1104.
- Wainwright J., Mulligan M. (2004) Environmental modelling, Wiley & Sons, Chichester.
- Wallace D.R., Fujii R.U. (1989) Software verification and validation: an overview, IEEE Software 6, 10–17.
- Wallach D. (2006) Evaluating crop models, in: Wallach D., Makowski D., Jones J.W. (Eds.), Working with dynamic crop models, Elsevier, Amsterdam, The Netherlands, pp. 11–53.
- Wallach D., Goffinet B. (1989) Mean square error of prediction in models for studying ecological and agronomic systems, Biometrics 43, 561–573.
- Westrich B. (2008) Model based sediment quality management on river basin scale, in: Sànchez-Marrè M., Béjar J., Comas J., Rizzoli A.E., Guariso G. (Eds.), Integrating sciences and information technology for environmental

assessment and decision making, Proc. 4th Biennial Meeting of the International Environmental Modelling and Software Society, 7–10 July, Barcelona, Spain, Vol. 1, pp. 639–646.

- Whitmore A.P. (1991) A method for assessing the goodness of computer simulations of soil processes, J. Soil Sci. 42, 289–299.
- Willmott C.J. (1981) On the validation of models, Phys. Geogr. 2, 184–194.
- Willmott C.J. (1982) Some comments on the evaluation of model performance, Bull. Am. Meteorol. Soc. 63, 1309–1313.
- Woodward S.J.R. (2001) Validating a model that predicts daily growth and feed quality of New Zealand dairy pastures, Environ. Int. 27, 133–137.
- Wright S.A. (2001) Covalidation of dissimilarly structured models, Dissertation, Air Force Institute of Technology, Dayton, OH, USA.

- Wright G.G., Tapping J., Matthews K.B., Wright R. (2003) Combining metric aerial photography and near-infrared videography to define within-field soil sampling frameworks, GeoCarto International 18, 1–8.
- Yagow E.R. (1997) Auxiliary procedures for the AGNPS model in urban fringe watersheds. PhD-Thesis, Virginia Polytechnic Institute, Blacksburg, VA, USA.
- Yang J., Greenwood D.J., Rowell D.L., Wadsworth G.A., Burns I.G. (2000) Statistical methods for evaluating a crop nitrogen simulation model, N-ABLE, Agr. Syst. 64, 37–53.
- Zacharias S., Coakley C.W. (1993) Comparison of quantitative techniques used for pesticide model validation, American Society of Agricultural Engineers. St. Joseph, MI, USA, ASAE Paper No. 93-2506.
- Zacharias S., Heatwole C.D., Coakley C.W. (1996) Robust quantitative techniques for validating pesticide transport models, Trans. ASAE 39, 47–54.

# **Cold Stress Tolerance Mechanisms in Plants**

Sudesh Kumar Yadav

## Contents

1	Introduction	606	
2	Effect of Cold on Plant Physiology	607	
3	Sensing and Signal Transduction During Cold Stress	609	
4	Role of Cold-Responsive Genes and Transcription Factors	611	
5	Conclusion	615	
<b>References</b>			

S.K. Yadav (🖂)

Abstract The human population is increasing at an alarming rate, whereas at the same time agricultural productivity is decreasing due to the effect of various environmental problems. In particular, cold stress is a serious threat to the sustainability of crop yields. Indeed, cold stress can lead to major crop losses. Various phenotypic symptoms in response to cold stress include poor germination, stunted seedlings, yellowing of leaves (chlorosis), reduced leaf expansion and wilting, and may lead to death of tissue (necrosis). Cold stress also severely hampers the reproductive development of plants. The major negative effect of cold stress is that it induces severe membrane damage. This damage is largely due to the acute dehydration associated with freezing during cold stress. Cold stress is perceived by the receptor at the cell membrane. Then a signal is transduced to switch on the cold-responsive genes and transcription factors for mediating stress tolerance. Understanding the mechanism of cold stress tolerance and genes involved in the cold stress signaling network is important for crop improvement. Here, I review cold stress tolerance mechanisms in plants. The major points discussed are the following: (1) physiological effects of cold stress, (2) sensing of cold temperatures and signal transduction, and (3) the role of various cold-responsive genes and transcription factors in the mechanism of cold stress tolerance.

**Keywords** Cold stress • Signal transduction • Coldresponsive genes • Transcription factors

Biotechnology Division, Institute of Himalayan Bioresource Technology, CSIR, Palampur-176 061 (HP), India e-mail: skyt@rediffmail.com; sudeshkumar@ihbt.res.in

## 1 Introduction

One-third of the total land area is considered as potentially suitable for arable agriculture. However, only some 10% of the world's 13 billion hectares are farmed. This is because abiotic stress in one form or another limits production on most of the world's 1.4 billion farmed hectares of land. This is a problem that is not going away. Among abiotic stresses, low and high temperature stress is very critical for determining the agricultural production. Plants exhibit a maximum rate of growth and development at an optimum temperature or over a diurnal range of temperatures (Fitter and Hay, 1981). When ambient temperature deviates from optimal, physiological, biochemical, metabolic and molecular changes occur within plants. This is an effort of plants to maximize growth and developmental processes and to maintain cellular homeostasis during such adverse conditions. Under increasingly stressful conditions, plants experience progressively more abnormal, impaired or dysfunctional cellular and whole-plant processes until the cardinal temperatures for survival are reached (Fitter and Hay, 1981). At the extremes of the natural temperature range of a plant, the degree of physiological, cellular, metabolic and molecular dysfunction becomes so severe that it leads to death.

Plants feel stress under high as well as low temperature exposure. Mesophilic temperate plants possess inducible temperature-stress tolerance. During high temperature exposure, stress tolerance can be induced by exposure to short-term elevated temperature and this is known as acquired thermotolerance (Kotak et al., 2007), while at lower temperatures, stress tolerance can be induced by exposure to reduced temperature and is known as chilling tolerance and/or cold acclimation. Chilling tolerance is the ability of a plant to tolerate low temperatures  $(0-15^{\circ}C)$  without injury or damage (Somerville, 1995), while cold acclimation is an enhanced tolerance to the physical and physiochemical vagaries of freezing stress (Guy, 1990; Thomashow, 1999). Both cold acclimation and chilling tolerance involve an array of biochemical, molecular and metabolic processes (Thomashow, 1999; Larkindale et al., 2005; Kotak et al., 2007; Zhu et al., 2007).

Exposure of plants to temperature stress leads to the modification of metabolism in two ways. First, plants try to adjust their cellular metabolism that altered due to rising or falling of temperatures. Temperature stress changes the structure, catalytic properties and function of enzymes (Kubien et al., 2003) and membrane metabolite transporters. Interestingly, regulatory mechanisms of plants become active and function to restore normal metabolite levels, and most importantly, metabolic fluxes (Schwender et al., 2004; Fernie et al., 2005). Secondly, the modifications of metabolism in response to temperature stress are mainly linked to enhanced tolerance mechanisms. Many metabolites thought to have important properties that could contribute to induced stress tolerance have long been linked to stress responses (Guy, 1990; Thomashow, 1999; Nayyar et al., 2005). Particular interest has been focused on metabolites that can function as osmolytes. Osmolytes are involved in the regulation of cellular water relations and reduce cellular dehydration. Their compatible solute behavior allows them to function to stabilize enzymes, membranes and other cellular components. Osmolytes are also involved in retailoring of membrane lipid composition to optimize the liquid/crystalline physical structure necessary for proper membrane function and energy sources. Such stress-responsive metabolites particularly include soluble sugars, amino acids, organic acids, polyamines and lipids (Levitt, 1972; Guy, 1990; Nayyar et al., 2005; Farooq et al., 2009).

Plants experience cold or chilling stress at temperatures from 0–15 °C. Under such situations, plants try to maintain homeostasis to acquire freezing tolerance and this involves extensive reprograming of gene expression and metabolism (Thomashow, 1999; Cook et al., 2004). In the recent past, great attention has been paid towards elucidation of the intricate signal transduction pathways responsible for low-temperature response (Lee et al., 2005). Fundamental responses of plants during cold stress exposure and acclimation mechanisms are presented in Fig. 1. Through work carried out in the last decade or so, several key components of signaling pathways and regulatory mechanisms contributing to freezing tolerance have been identified. This article summarizes the latest work that addresses the following questions: what are the physiological effects of cold stress in plants? How are cold temperatures sensed by plants? How is a cold stress signal transduced to the nucleus to regulate gene expression? What are different genes and transcription factors involved in the mechanism of low temperature tolerance?



**Fig. 1** Fundamental responses of plants during cold stress exposure. Cold stress exposure causes various physiochemical disturbances, leading to growth inhibition. Cold stress response is perceived by plants through a signal transduction that leads to the activation of transcription factors and cold-responsive genes. Such transcription factors and genes control the damage due to cold stress and help in providing tolerance to plants

## 2 Effect of Cold on Plant Physiology

Each plant has an optimum set of temperatures for its proper growth and development. A particular set of temperature conditions, which are optimum for one plant, may be stressful for another plant. It has generally been noticed that plants native to warm habitats exhibit symptoms of injury upon exposure to low non-freezing temperatures (Lynch, 1990). For example, plants such as maize (*Zea mays*), soybean (*Glycine max*), cotton (*Gossypium hirsutum*), tomato (*Lycopersicon esculentum*) and banana (*Musa* sp.) show signs of injury upon exposure to temperatures below 10–15 °C (Lynch, 1990; Guy, 1990; Hopkins, 1999). However, the appearance of injury symptoms depends upon the sensitivity of a plant to cold stress and varies from plant to plant. Cold stress-induced injury in plants may appear after 48 to 72 h of stress exposure. Plants exposed to cold stress show various phenotypic symptoms that include reduced leaf expansion, wilting and chlorosis (yellowing of leaves) and may lead to necrosis (death of tissue). The phenotypic symptoms of rice seedlings upon exposure to cold stress are shown in Fig. 2. Cold stress also severely affects the reproductive development of plants and this has been seen in rice plants at the time of anthesis (floral opening), which leads to sterility in flowers (Jiang et al., 2002).

Cold stress generally results in poor germination, stunted seedlings, yellowing of leaves, withering and reduced tillering. The effects of cold stress at the reproductive stage of plants delay heading and result in pollen sterility, which is thought to be one of the key factors responsible for the reduction in grain yield of crops (Suzuki et al., 2008). The major adverse effect of cold stress in plants has been seen in terms of plasma membrane damage. This has been documented due to cold stress-induced dehydration (Steponkus, 1984; Steponkus et al., 1993). The plasma membrane is made up of lipids and proteins. Lipids in the plasma membrane are made up of two kinds of fatty acids: unsaturated and saturated fatty acids. Unsaturated fatty acids have one or more double bonds between two carbon atoms, whereas saturated fatty acids are fully saturated with hydrogen atoms. Lipids containing more saturated fatty acids solidify faster and at temperatures higher than those containing unsaturated fatty acids. Therefore, the relative proportion of these two types of fatty acids in the lipids of the plasma membrane determines the fluidity of the membrane (Steponkus et al., 1993). At the transition temperature, a membrane changes from a semi-fluid state into a semi-crystalline state. Cold-sensitive plants usually have a higher proportion of saturated fatty acids in their plasma membrane. Therefore, cold-sensitive plants have a higher transition temperature. On the contrary, cold-resistant plants have a higher proportion of unsaturated fatty acids and hence a lower transition temperature.

The agricultural crops which can withstand even during the freezing temperatures of late spring or early fall frost can be used more successfully for cultivation during cold stress. Therefore, selection of low temperature-tolerant crops is very important for the sustainability of agriculture. Additionally, understanding of how cold stress induces its injurious effects on plants is crucial for the development of frost-tolerant



**Fig. 2** Effect of cold stress on rice seedlings. Rice seedlings exposed to  $-2^{\circ}$ C for 12 h and then allowed to recover for 1 week. Damage due to cold stress is seen as retardation in growth, reduced leaf expansion, wilting, chlorosis (yellowing of leaves)

and necrosis (death of tissue). Picture on right hand side presents a closer view of the unstressed and cold stress-exposed leaves of rice

crops. It has been noticed that cold-induced ice formation is the real cause of plant damage. Ice formation in plant tissues during cold stress leads to dehydration. Ice is formed in the apoplastic space of a plant tissue because that has relatively lower solute concentration. It is known that the vapor pressure of ice is much lower than water at any given temperature. Therefore, ice formation in the apoplast establishes a vapor pressure gradient between the apoplast and surrounding cells. As a result of this gradient, the cytoplasmic water migrates down the gradient from the cell cytosol to the apoplastic space. This adds to existing ice crystals in the apoplastic space and causes a mechanical strain on the cell wall and plasma membrane, leading to cell rupture (McKersie and Bowley, 1997; Olien and Smith, 1997; Uemura and Steponkus, 1997).

In addition to the well-established harmful effect of cold stress alterations in lipid composition of the biomembranes, affecting their fluidity (Senser and Beck, 1982; Quinn, 1985; Williams, 1990; Welti et al., 2002), certain additional factors may also contribute to damage induced by cold stress. This includes synthesis and accumulation of compatible solutes, the synthesis of cold acclimation-induced proteins (Close, 1997; Shinozaki and Yamaguchi-Shinozaki, 2000), changes in the carbohydrate metabolism (Hansen and Beck, 1994; Hansen et al., 1997; Liu et al., 1998; Frankow-Lindberg, 2001) and the boosting of the radical scavenging potential of the cells (Tao et al., 1998; Hernández-Nistal et al., 2002; Baek and Skinner, 2003). Taken together, cold stress results in loss of membrane integrity, leading to solute leakage. Further, cold stress disrupts the integrity of intracellular organelles, leading to the loss of compartmentalization. Exposure of plants to cold stress also causes reduction and impairing of photosynthesis, protein assembly and general metabolic processes.

Recently, attempts have also been directed towards analyzing the effect of cold stress on the whole-plant metabolome. Metabolic profiling of Arabidopsis plants revealed that cold acclimation increases 75% of the 434 analyzed metabolites (Cook et al., 2004; Kaplan et al., 2004). However, metabolite profiles in Arabidopsis do not appear to correlate with cold acclimation capacity (Hannah et al., 2006). The role of such metabolites in plants has been known as osmoprotectants. In addition to their role as osmoprotectants and osmolytes, certain metabolites (individual metabolites or the redox state) induced during cold acclimation might act as signals for reconfiguring gene expression. For example, cold stress induces the accumulation of proline, a well-known osmoprotectant. Microarray and RNA gel blot analyses have shown that proline can induce the expression of many genes, which have the proline-responsive element sequence ACTCAT in their promoters (Satoh et al., 2002; Oono et al., 2003).

Cold stress affects virtually all aspects of cellular function in plants. One of the major influences of cold stress-induced dehydration is membrane disintegration. Such changes caused by cold stress adversely affect the growth and development of plants.

# 3 Sensing and Signal Transduction During Cold Stress

Plants are sessile in nature. Therefore, the only way for them to survive under adverse environmental conditions is to adapt to changing surroundings quickly and efficiently. The plastic nature of higher plants makes them able to react to different stresses with specific responses in growth, development and metabolism. In view of this, understanding of such plant responses could be of great importance for agriculture. This might be helpful in developing stress-tolerant crop varieties. Cold stress exposure reduces the fluidic nature of cellular membranes and increases their rigidity. This documents that the primary site of cold stress sense in plants could be associated with membrane fluidity, protein and nucleic acid conformation, metabolite concentration, and cellular redox status.

Cold stress is sensed by plant cells through its membrane rigidification effect. Rigidification of the plasma membrane during cold exposure has been shown to induce cold-responsive genes that help in cold acclimation in alfalfa and Brassica napus (Orvar et al., 2000; Sangwan et al., 2001). The Arabidopsis fad2 mutant defective in oleate desaturase exhibits membrane rigidification and cold stress sensitivity, suggesting the relation of cold stress sense to plasma membrane rigidification (Vaultier et al., 2006). Additionally, cold stress exposure in plants has been found to increase cytosolic calcium levels. This increase in calcium is mediated through membrane rigidification-activated mechano-sensitive or ligand-activated calcium channels. The higher level of calcium in the cytosol leads to signal amplification through phospholipids (Vergnolle et al., 2005; Williams et al., 2005; Chinnusamy et al., 2006; Komatsu et al., 2007).

Environmental stresses including cold stress are first perceived by the receptors present on the membrane of the plant cells. The specific receptor for cold stress is not yet known. The signal is then transduced downstream and many signaling pathways are activated. Studies have shown that such pathways are often activated in concert. The various components are calcium, reactive oxygen species, protein kinase, protein phosphatase and lipid signaling cascades. It is believed that specificity is achieved by the combination and timing of the activation of different signaling pathways. The change in cytosolic calcium level is sensed by calcium-binding proteins. These calcium-binding proteins do not possess enzymatic activity but undergo conformation changes in a calcium-dependent manner. The change in calcium-binding proteins makes them interact with other proteins and often initiates a phosphorylation cascade. Through this cascade, plant cells could target major stress-responsive genes or the transcription factors. Transcription factors also regulate the expression and function of genes, which ultimately leads to plant adaptation and survival during unfavorable conditions (Shinozaki and Yamaguchi-Shinozaki, 2000; Shinozaki et al., 2003; Mahajan and Tuteja, 2005; Yamaguchi-Shinozaki and Shinozaki, 2006).

Individual plant cells respond to the environmental stress in the way described above and then the whole plant acts synergistically. The change in gene expression governed by the signal cascade mechanism also induces changes in genes participating in the formation of plant hormones such as abscisic acid, salicylic acid and ethylene. These hormones may amplify the same cascade or may initiate a new signaling pathway. Additionally, several other cellular components are also involved in the stress signal transduction mechanism. These accessory molecules may not directly participate in signaling but participate in the modification or assembly of major signaling components. Mainly such components are protein modifiers and act in post-translational modification of signaling proteins. Such modifiers are involved in myristoylation, glycosylation, methylation and ubiquitination of signaling proteins (Mahajan and Tuteja, 2005; Yamaguchi-Shinozaki and Shinozaki, 2006).

Lipid molecules are also very important in signal transduction during cold stress. Though lipid signaling is relatively less studied, phosphatidic acid produced by both phospholipase D and the concerted action of phospholipase C and diacylglycerol kinase has been proposed as a membranous secondary messenger molecule. Phosphatidic acid is rapidly and transiently generated in response to various stresses and has been proposed to function as a second messenger (Meijer and Munnik, 2003). This phosphatidic acid constitutes a minor portion of membrane lipids under control conditions, but its levels significantly
increased upon exposure to numerous stresses including cold stress (Munnik, 2001; Meijer and Munnik, 2003). Several modes of action for phosphatidic acid in signal transduction can be imagined. Its functions are mainly based on the recruitment and regulation of enzymatic activity of target proteins. The recruitment of soluble proteins to particular membranes can have numerous effects, such as localization of a protein to the site where it is active, localization of a protein to a site where it is modified and sequestration of a protein from the site where it is active. Alternatively, phosphatidic acid could influence the enzyme activity of proteins already residing in the membrane or of recruited proteins or protein complexes. A plethora of phosphatidic acid-binding proteins have been found in different organisms, and many have been proposed to function in signaling cascades (Testerink and Munnik, 2005).

Abscisic acid is an important phytohormone that plays a crucial role in several plant stress responses, including cold stress (Verslues and Zhu, 2005; Mahajan and Tuteja, 2005). Furthermore, phospholipase D has been linked to reactive oxygen species, which are known to be involved in abscisic acid and in cold stress responses (Laloi et al., 2004; Mahajan and Tuteja, 2005). In Arabidopsis, different phospholipases D such as AtPLDa1 and AtPLDd have been implicated in both the production of and responses to reactive oxygen species (Zhang et al., 2003, 2005). The AtPLDd expression was shown to be induced by abscisic acid and cold in Arabidopsis plants (Katagiri et al., 2001). Coldinduced freezing tolerance is reportedly impaired in Atpldd T-DNA knock-out Arabidopsis mutant plants and enhanced in AtPLDd-overexpressing plants, which also display decreased and increased freezing-induced phosphatidic acid production, respectively (Li et al., 2004). As a whole, it seems that phospholipases D are involved in multiple aspects of both the overlapping and distinct signaling networks that are activated by cold stress. Cold stress-induced increase in phospholipases D is also involved in changes in membrane fluidity and osmotic balance.

Secondary signals, such as abscisic acid and reactive oxygen species, can also induce calcium signatures that impact cold signaling. *Arabidopsis* mutants defective in the activation of the molybdenum cofactor of abscisic aldehyde oxidase, namely aba3/freezing sensitive 1 (*frs1*) (Llorente et al., 2000), also known as *los5* (low expression of osmotically responsive genes 5) (Xiong et al., 2001), exhibit hypersensitivity to freezing stress. The los5 mutant plants show a significant reduction in the expression of cold- and osmotic stressinduced genes (Xiong et al., 2001). Reactive oxygen species accumulate in plant cells during exposure to various abiotic stresses, and they appear to have a strong influence on cold regulation of gene expression. The Arabidopsis fro1 (frostbite1) mutant, which constitutively accumulates high levels of reactive oxygen species, exhibits impaired expression of coldresponsive genes and hypersensitivity to chilling and freezing. The FRO1 gene encodes the Fe-S subunit of complex I (NADH dehydrogenase) of the respiratory electron transfer chain in mitochondria, and its disruption leads to high levels of reactive oxygen species generation (Lee et al., 2002). Besides their effect on calcium signatures, reactive oxygen species signals can also exert their effects directly through the activation of redox-responsive proteins, such as transcription factors and protein kinases.

Ultimately, cold acclimation involves precise regulation of expression of transcription factors and effector genes, collectively known as cold-regulated genes (Thomashow, 1999; Viswanathan and Zhu, 2002). Significant progress has been made in identifying transcriptional, post-transcriptional and post-translational regulators of cold-induced expression of cold-regulated genes. Promoters of many of the cold-regulated genes contain cis-elements such as dehydration-responsive elements/C-repeat elements (A/GCCGAC), an abscisic acid-responsive element (PyACGTGGC), and myeloblastosis recognition sequences (C/TAACNA/G) and/or myelocytomatosis recognition sequences (CANNTG) (Yamaguchi-Shinozaki and Shinozaki, 2005). Additionally, abiotic stresses including cold stress regulate the expression and activity of various kinases of mitogenactivated protein kinase pathways. Under cold stress, reactive oxygen species activate the AtMEKK1/ ANP1 (MAPKKK)-AtMKK2 (MAPKK)-AtMPK4/6 (MAPK) mitogen-activated protein kinase cascade that is necessary for cold acclimation in plants (Kovtun et al., 2000; Teige et al., 2004). This suggests that mitogen-activated protein kinase cascades act as a converging point in abiotic stress signaling.

Cold stress is sensed by a yet unknown receptor. The signal is then transduced through several components of signal transduction pathways. The components of signal transduction pathways discussed here are also involved in the transduction of other environmental stress signals, suggesting a common pathway for several responses in plants. The cold stress signal leads to changes in the expression of coldregulated genes and the level of expression determines the fate of plants under such stress.

## 4 Role of Cold-Responsive Genes and Transcription Factors

Developing transgenic plants for cold tolerance is a fast and effective biotechnological tool to improve agricultural crops. Transgenic approaches have been used to improve cold tolerance in rice. Various studies on crop improvement have indicated that screening for genes involved in cold tolerance is a crucial initial step for crop improvement strategy using engineering (Hsieh et al., 2002; Dubouzet et al., 2003; Ohnishi et al., 2005; Ito et al., 2006). Among the genes identified for cold tolerance, genes encoding transcription factors are found to be better for improving stress tolerance in plants. Most of such transcription factors have been identified in Arabidopsis and rice. The functional and biochemical features of specific cell types are determined by their particular gene expression profiles. Such global gene expression patterns can be represented by a "transcriptome", which reveals the identity and the level of expression of each expressed gene in a defined population of cells (Velculescu et al., 1997; Tuteja and Tuteja, 2004). The transcriptome can be modulated by both external and internal factors, and thereby provide not only a wealth of basic biological insights but also a global view of biological responses to environmental stimuli.

Transcriptome expression profiles of a plant under normal and stressed conditions can be obtained and compared by various methods, such as ribonucleic acid (RNA)–deoxyribonucleic acid (DNA) hybridization measurements, subtractive hybridization, subtraction libraries and differential display (Donson et al., 2002). However, these methods have technical shortcomings and therefore, are unable to provide overall gene expression patterns. The recent DNA microarray technique allows larger-scale quantitative gene expression analysis. Microarray is a very useful technique for transcriptome analysis. This has been used in identifying many stress-inducible genes involved in stress response and tolerance (Shinozaki et al., 2003; Seki et al., 2004). A large number of genes responsive to various abiotic stresses have been identified using various microarrays (Seki et al., 2001, 2002; Fowler and Thomashow, 2002; Kreps et al., 2002; Rabbani et al., 2003; Bray, 2004; Maruyama et al., 2004; Vogel et al., 2005). However, microarray has some innate limitations. It only allows the analysis of arbitrarily chosen genes (Duggan et al., 1999; Jones et al., 2001; Patankar et al., 2001; Lorenz and Dean, 2002; Gibbings et al., 2003). Another technology known as serial analysis of gene expression partially overcomes this limitation (Velculescu et al., 1995). It is an extremely promising, efficient and global approach for analyzing gene expression profiles under different physiological states. It has the advantage over various other techniques that it allows identifying novel genes along with their associated pathways and metabolic circuits. Also, the Affymetrix 22K GeneChip ATH1 has been used recently to identify more stress-inducible genes. The genes identified using the Affymetrix 22K GeneChip ATH1 can be seen at The Arabidopsis Information Resource Uniform Resource Locator (http://www.arabidopsis.org/).

Upon stress exposure, the expression of various genes alters. Such stress-responsive genes have been found to provide protection to plants in two ways. They protect from stress by producing important metabolic proteins. The products of stress-responsive genes are also involved in regulation of genes of signal transduction pathways. Based upon different functions of proteins encoded by stress-responsive genes, these are classified into two groups (Fowler and Thomashow, 2002; Kreps et al., 2002; Seki et al., 2002). The first group includes proteins that probably function in stress tolerance, such as chaperones, late embryogenesis abundant proteins, osmotin, antifreeze proteins, mRNA-binding proteins, key enzymes for osmolyte biosynthesis such as proline, water channel proteins, sugar and proline transporters, detoxification enzymes, enzymes for fatty acid metabolism, proteinase inhibitors, ferritin, and lipid-transfer proteins. Some of these stress-inducible genes encoding key enzymatic proteins for osmolyte biosynthesis, late embryogenesis abundant proteins, and detoxification enzymes have been overexpressed in transgenic plants. Such transgenics have been found to be stress-tolerant (Cushman and Bohnert, 2000). This has evinced that the gene products of the stress-inducible genes really function in stress tolerance.

The second group contains protein factors involved in further regulation of signal transduction and gene expression that probably function in stress response. This group includes various transcription factors (Seki et al., 2003). These transcription factors regulate various stress-inducible genes cooperatively or separately, and may constitute gene networks. Functional analysis of such stress-inducible transcription factors should provide more information on the complex regulatory gene networks that are involved in responses to stresses including cold stress. There are a few more proteins involved in the complex network of stress regulatory mechanism of plants such as kinases, protein phosphatases, enzymes involved in phospholipid metabolism, calmodulin binding protein and 14-3-3 proteins. The functions of most of these genes are not fully understood yet. Some of these stress-inducible regulatory genes that encode transcription factors have been overexpressed in plants and generated stresstolerant phenotypes in transgenic plants (Zhang et al., 2004; Tester and Bacic, 2005; Vinocur and Altman, 2005).

Cold stress-induced genes are known as coldresponsive genes. Most of such genes encoding polypeptides are homologs of late embryogenesis abundant proteins and the polypeptides that are synthesized during the late embryogenesis phase (Dure, 1993; Ingram and Bartels, 1996; Close, 1997). These late embryogenesis abundant-like proteins are mainly hydrophilic and have relatively simple amino-acid composition. They are composed largely of a few amino acids with repeated amino acid sequence motifs and are predicted to contain regions capable of forming amphipathic alpha-helices. The examples of cold-responsive genes include COR15a (Artus et al., 1996), alfalfa Cas15 (Monroy et al., 1993) and wheat WCS120 (Houde et al., 1992). The expression of cold-responsive genes has been shown to be critical for both chilling tolerance and cold acclimation in plants. The identified Arabidopsis cold-responsive genes are COR78/RD29, COR47, COR15a and COR6.6 (Thomashow, 1999). These genes are induced by cold, dehydration or abscisic acid. A COR15A polypeptide encoded by the COR15a gene is targeted to the chloroplast. Formation of hexagonal II phase lipids is a major cause of membrane damage in non-acclimated plants during cold stress. However, COR15a expression decreases the propensity of the membranes to form hexagonal II phase

lipids in response to freezing (Uemura and Steponkus, 1997). This observation suggests the role of COR15A polypeptide in providing freezing tolerance.

Every gene has promoter sequences upstream of their coding region. The expression of a gene is driven by its promoter. Promoters have regulatory elements in their sequences that control the expression of a gene. In view of this, analysis of the promoter elements of cold-responsive genes revealed that they contain dehydration-responsive elements or C-repeats. Some of the cold-responsive genes also contain an abscisic acid-responsive element in their promoter sequences (Yamaguchi-Shinozaki and Shinozaki, 1994; Stockinge et al., 1997). The expression of coldresponsive genes is induced during cold exposure. Exposure of plants to cold stress also induces the expression of a transcription factor, i.e. dehydrationresponsive elements or C-repeat binding factors. This suggests that these binding factors regulate the expression of cold-responsive genes during cold stress. Overexpression of dehydration-responsive elements or C-repeat binding factors led to the induction of coldresponsive genes under unstressed conditions, documenting the regulatory role of binding factors in coldresponsive gene expression (Stockinge et al., 1997). Cold stress induces three dehydration-responsive elements or C-repeat binding factor genes in Arabidopsis such as AtCBF1 (DREB1B), AtCBF2 (DREB1C) and AtCBF3 (DREB1A). Dehydration-responsive elements or C-repeat binding factors bind to the promoter of cold-responsive genes and other cold-regulated genes. The overexpression of these regulatory elements not only resulted in increased freezing tolerance but also an increase in drought tolerance (Liu et al., 1998). This finding provides strong support that a fundamental role of cold-responsive genes is to protect the plant cells against cellular dehydration.

The role of these various factors has been substantiated by developing transgenics. Constitutive or stressinducible overexpression of dehydration-responsive elements or C-repeat binding factor-1 or -3 in transgenic *Arabidopsis* plants resulted in constitutive or enhanced expression of cold-responsive genes. This increased abiotic-stress tolerance, including freezing tolerance, in *Arabidopsis* (Jaglo-Ottosen et al., 1998; Liu et al., 1998; Kasuga et al., 1999). Overexpression of *Arabidopsis thaliana* dehydration-responsive elements or C-repeat binding factor-1 or -3 also enhanced chilling, freezing, drought and/or salt-stress

tolerance in Brassica (Jaglo et al., 2001), tomato (Hsieh et al., 2002), tobacco (Kasuga et al., 2004), wheat (Pellegrineschi et al., 2004) and rice (Oh et al., 2005). Similarly, overexpression of rice (Dubouzet et al., 2003) and maize (Qin et al., 2004) dehydrationresponsive element binding factor-1 in transgenic Arabidopsis induced constitutive expression of coldresponsive genes and conferred tolerance to freezing and drought stress. These studies have documented that dehydration-responsive elements or Crepeat binding factor-dependent gene expression is an important, evolutionarily conserved component of cold acclimation in diverse plant species. However, transcriptome analysis of dehydration-responsive elements or C-repeat binding factors overexpressing transgenic Arabidopsis revealed that only  $\sim 12\%$  of the cold-responsive genes are induced by dehydrationresponsive elements or C-repeat binding factors (Fowler and Thomashow, 2002). This suggests the possibility of other transcriptional activators' or repressors' role in cold acclimation.

To find this, T-DNA insertion mutants of Arabidopsis were screened. Genetic analysis of mutants led to the identification of an inducer of dehydrationresponsive elements or C-repeat binding factor expression-1 as an activator of dehydration-responsive elements or C-repeat binding factor-3 (Chinnusamy et al., 2003). The inducer of dehydration-responsive elements or C-repeat binding factor expression-1 encoded a transcription factor that specifically recognized myelocytomatosis sequence on the promoter of dehydration-responsive elements or C-repeat binding factor-3. Interestingly, transgenic Arabidopsis overexpressing an inducer of dehydration-responsive elements or C-repeat binding factor expression-1 did not express dehydration-responsive elements or C-repeat binding factor-3 at warm temperatures, but showed a higher level of expression for dehydrationresponsive elements or C-repeat binding factor-3 as well as other cold-responsive genes such as RD29 and COR15a at low temperatures. This suggests that cold stress induced an inducer of dehydration-responsive elements or C-repeat binding factor expression-1 is a crucial activator of dehydration-responsive elements or C-repeat binding factor-3 in plants. In addition to the role of an inducer of dehydration-responsive elements or C-repeat binding factor expression-1 in regulating dehydration-responsive elements or C-repeat binding factors, homeodomain leucine zipper

has also been found to interact with dehydrationresponsive elements or C-repeat binding factors. The expression of two dehydration-responsive elements or C-repeat binding factor-1 such as cDNAs CaCBFIA and CaCBFIB in hot pepper was induced in response to low temperature stress (4 °C) (Kim et al., 2004). The induction in expression of homeodomain leucine zipper and its interaction with CaCBFIB during cold stress suggested that homeodomain leucine zipper could also be an important regulon of dehydrationresponsive elements or C-repeat binding factors. So far, studies have evinced the existence of one of the major and important molecular mechanisms of cold tolerance in plants, as presented in Fig. 3.

One of the largest families of transcription factors, myeloblasts also play a very significant role in regulating gene expression in response to environmental and developmental changes. In this category, myeloblasts are predominant transcription factors. Based on the number of tandem repeats of SANT 'SWI3, ADA2, N-CoR and TFIIIB' DNA-binding domains myeloblastosis transcription factors are divided into three subfamilies: myeloblastosis-like proteins (MYB1R factors), R2R3-type myeloblastosis factors and R1R2R3 myeloblastosis (MYB3R) factors with one, two or three repeats, respectively (Rosinski and Atchley, 1998).

In Arabidopsis, 198 genes encode myeloblastosis transcription factors and it is one of the largest families among all of the transcription factor families (Yanhui et al., 2006; Pasquali et al., 2008). The function of some of these myeloblastosis transcription factors has been elucidated through T-DNA insertional mutagenesis and transcript expression profiling analysis (Kranz et al., 2000; Stracke et al., 2001; Yanhui et al., 2006). This has revealed the roles of R2R3 myeloblastosis proteins in cell cycle control, secondary metabolism, cellular morphogenesis, meristem formation, hormonal signaling and stress responses (Salomoni et al., 1997; Meissner et al., 1999). Several R2R3 myeloblastosis genes have been found to be involved in the responses to environmental stimuli including cold stress (Yanhui et al., 2006). In this category, high expression of osmotically responsive genes such as HOS10 encoding a R2R3-type myeloblastosis transcription factor is essential for cold acclimation (Zhu et al., 2005). A myeloblastosis 15 controls the expression of dehydration-responsive elements or C-repeat binding factors and other cold-responsive



614

**Fig. 3** Involvement of various transcription factors in the induction of cold-responsive genes during cold stress. Cold stress exposure in plants induces the calcium signature. This signature then induces an inducer of dehydration-responsive elements or C-repeat binding factor expression-1 in its inactive form. The variant ring finger protein high expression of osmotically responsive gene-1 has been identified as a negative regulator of cold responses. The high expression of osmotically responsive gene-1 is an E3 ligase required for the ubiquitination of an inducer of dehydration-responsive elements or C-repeat binding factor expression-1. High expression of the osmotically responsive gene-1 physically interacts with an inducer of dehydration-responsive elements or C-repeat binding factor expression-1 and mediates the proteolysis and phos-

phorylation of an inducer of dehydration-responsive elements or C-repeat binding factor expression-1, converting it from the inactive to active form. An active inducer of dehydrationresponsive elements or C-repeat binding factor expression-1 regulates the expression of dehydration-responsive elements or Crepeat binding factors and other transcription factors by binding to myelocytomatosis recognition sequences in their promoter regions. The dehydration-responsive elements or C-repeat binding factors also regulate the expression of other transcription factors. These various transcription factors then regulate the expression of various cold-responsive genes by acting on cis-elements and C-repeats/dehydration-responsive elements in their promoter or upstream sequences. Induction in expression of cold-responsive genes determines tolerance during cold stress genes (Agarwal et al., 2006). Overexpression of rice myeloblastosis transcription factors *OsMYB3R-2* and *OsMYB4* in *Arabidopsis* has been found to significantly increase the cold tolerance (Vannini et al., 2004; Dai et al., 2007; Pasquali et al., 2008).

In addition to the role of late embryogenesis abundant-like proteins and myeloblastosis transcription factors in cold acclimation, recent studies have found the essential role of the Zinc-finger group of proteins in cold stress tolerance. Overexpression of Zincfinger genes such as *OsiSAP8*, *OsCOIN*, *OsISAP1 OsbHLH1*, *OsDREB1/CBF*, *ROs-bZIP*, *SNAC2* and *OsNAC6* confers cold stress tolerance at the seedling stage in rice (Wang et al., 2003; Mukhopadhyay et al., 2004; Ohnishi et al., 2005; Ito et al., 2006; Cheng et al., 2007; Liu et al., 2007; Nakashima et al., 2007; Hu et al., 2008; Kanneganti and Gupta, 2008).

The use of advanced biotechnological tools has enabled us to identify the role of several other signaling components and metabolic regulators in stress response. A OsTPP1 gene encoding trehalose-6phosphate phosphatase has been documented to confer cold stress tolerance in rice by inducing the expression of cold-responsive genes (Pramanik and Imai, 2005; Shima et al., 2007; Ge et al., 2008). Similarly, hydrophobic proteins encoded by OsLti6 genes (OsLti6a and OsLti6b) also enhanced tolerance to cold stress in rice seedlings (Morsy et al., 2005). The Arabidopsis FAD8 gene (Gibson et al., 1994) encodes a fatty acid desaturase that contributes to freezing tolerance by altering the lipid composition. The cold-responsive genes encoding molecular chaperones, including a spinach heat shock protein gene hsp70 (Anderson et al., 1994), and a Brassica napus heat shock protein gene hsp90 (Krishna et al., 1995), contribute to freezing tolerance by stabilizing proteins against freezeinduced denaturation.

Stress responses in plants are also mediated by the mitogen-activated protein kinase cascades as described under the signal transduction section. Overexpression of a rice mitogen-activated protein kinase *OsMAPK5* conferred tolerance to various environmental stresses including cold stress in rice seedlings (Xiong and Yang, 2003). Stress-responsive genes encoding calcineurin B-like protein-interacting protein kinases such as *OsCIPK03* and *OsCIPK12* also play important roles in conferring cold stress tolerance in rice (Xiang et al., 2007). The role of calcium-dependent protein kinases in signal transduction of plants has also

been well documented. The calcium-dependent protein kinases encoded by OsCDPK7 and OsCDPK13 have been found to be positive regulators for enhancing cold stress tolerance (Saijo et al., 2000, 2001; Wan et al., 2007; Wang et al., 2008). Induction in expression of a rice calcium-dependent protein kinase OsCDPK13 and accumulation of OsCDPK13 protein were upregulated in response to cold (Abbasi et al., 2004). A rice calcium-dependent protein kinase OsCDPK13 overexpressing transgenic rice had higher recovery rates following cold stress in comparison with non-transgenic rice. Cold-tolerant rice varieties exhibited higher expression of a rice calcium-dependent protein kinase OsCDPK13 than the cold-sensitive ones (Abbasi et al., 2004). These studies have documented the important role of calcium-dependent protein kinases in mediating tolerance in response to cold stress.

For acclimation to cold stress, plants involve precise signaling and regulation of the transcriptome. Among the various reported transcription factors, an inducer of dehydration-responsive elements or C-repeat binding factor expression-1 and dehydration-responsive elements or C-repeat binding factors play pivotal roles in the regulation of cold-responsive genes and providing cold tolerance to plants.

## **5** Conclusion

Cold stress is one of the major environmental stresses that limit agricultural crop productivity by affecting their quality and post-harvest life. Most temperate plants acquire chilling and freezing tolerance upon prior exposure to sublethal cold stress, a process called cold acclimation. Still, many agronomically important crops are incapable of cold acclimation. Cold stress affects virtually all aspects of cellular function in plants. Such changes caused by cold stress adversely affect the growth and development of plants. Cold stress is sensed by a yet unknown receptor. The cold stress signal is transduced through several components of signal transduction pathways. Major components are calcium, reactive oxygen species, protein kinase, protein phosphatase and lipid signaling cascades. Abscisic acid also mediates the response of cold stress. The cold stress signal leads to regulation of transcription factors and effector genes, collectively called coldregulated genes. The effector genes encoding proteins

under this category include chaperones, late embryogenesis abundant proteins, osmotin, antifreeze proteins, mRNA-binding proteins, key enzymes for osmolyte biosynthesis such as proline, water channel proteins, sugar and proline transporters, detoxification enzymes, enzymes for fatty acid metabolism, proteinase inhibitors, ferritin, and lipid-transfer proteins. The transcription factors involved during cold stress response are inducers of C-repeat binding factor expression-1, C-repeat binding factors, myeloblasts and mitogenactivated protein kinase. Analyses of the expression of cold-regulated genes indicate the presence of multiple signal transduction pathways between the initial stress signals and gene expression. Use of these genes and transcription factors in genetic modification of agricultural crops can improve cold tolerance and productivity.

Acknowledgements I am grateful to Dr. P. S. Ahuja, Director, IHBT, for his useful suggestions. I would like to acknowledge the Council of Scientific and Industrial Research (CSIR) and Department of Science and Technology (DST), Govt. of India, for their financial help with conducting research in my laboratory.

### References

- Abbasi F., Onodera H., Toki S., Tanaka H., Komatsu S. (2004) OsCDPK13, a calcium dependent protein kinase gene from rice, is induced by cold and gibberellin in rice leaf sheath, Plant Mol. Biol. 55, 541–552.
- Agarwal M., Hao Y., Kapoor A., Dong C.H., Fujii H., Zheng X., Zhu J.K. (2006) A R2R3 type MYB transcription factor is involved in the cold regulation of CBF genes and in acquired freezing tolerance, J. Biol. Chem. 281, 37636–37645.
- Anderson J.V., Li Q.B., Haskell D.W., Guy C.L. (1994) Structural organization of the spinach endoplasmic reticulum-luninal 70-kilodalton heat shock cognate gene and expression of 70-kilodalton heat shock genes during cold acclimation, Plant Physiol. 104, 1359–1370.
- Artus N.N., Uemura M., Steponkus P.L., Gilmour S.J., Lin C.T., Thomashow M.F. (1996) Constitutive expression of the cold regulated *Arabidopsis thaliana* COR 15a gene affects both chloroplast and protoplast freezing tolerance, Proc. Natl Acad. Sci. (USA) 93, 13404–13409.
- Baek K.H., Skinner D.Z. (2003) Alteration of antioxidant enzyme gene expression during cold acclimation of nearisogenic wheat lines, Plant Sci. 165, 1221–1227.
- Bray E.A. (2004) Genes commonly regulated by water-deficit stress in Arabidopsis thaliana, J. Exp. Bot. 55, 2331–2341.
- Cheng C., Yun K.Y., Ressom H.W., Mohanty B., Bajic V.B., Jia Y., Yun S.J., de los Reyes B.G. (2007) An early response regulatory cluster induced by low temperature and hydrogen peroxide in seedlings of chilling-tolerant japonica rice, BMC Genomics 8, 175.

- Chinnusamy V., Ohta M., Kanrar S., Lee B.H., Hong X., Agarwal M., Zhu J.K. (2003) ICE1: a regulator of coldinduced transcriptome and freezing tolerance in *Arabidopsis*, Gene. Dev. 17, 1043–1054.
- Chinnusamy V., Zhu J., Zhu J.K. (2006) Gene regulation during cold acclimation in plants, Physiol. Plant. 126, 52–61.
- Close T.J. (1997) Dehydrins: A commonality in the response of plants to dehydration and low temperature, Physiol. Plantarum 100, 291–296.
- Cook D., Fowler S., Fiehn O., Thomashow M.F. (2004) A prominent role for the CBF cold response pathway in configuring the low-temperature metabolome of *Arabidopsis*, Proc. Natl Acad. Sci. (USA) 101, 15243–15248.
- Cushman J.C., Bohnert H.J. (2000) Genome approaches to plant stress tolerance, Curr. Opin. Plant Biol. 3, 117–124.
- Dai X., Xu Y., Ma Q., Xu W., Wang T., Xue Y., Chong K. (2007) Overexpression of an R1R2R3 MYB gene, OsMYB3R-2, increases tolerance to freezing, drought, and salt stress in transgenic *Arabidopsis*, Plant Physiol. 143, 1739–1751.
- Donson J., Fang Y., Espiritu-Santo G., Xing W., Salazar A., Miyamoto S., Armendarez V., Volkmuth W. (2002) Comprehensive gene expression analysis by transcript profiling, Plant Mol. Biol. 48, 75–97.
- Dubouzet J.G., Sakuma Y., Ito Y., Kasuga M., Dubouzet E.G., Miura S., Seki M., Shinozaki K., Yamaguchi-Shinozaki K. (2003) OsDREB genes in rice, *Oryza sativa* L., encode transcription activators that function in drought-, high- altand cold-responsive gene expression, Plant J. 33, 751–763.
- Duggan D.J., Bittner M., Chen Y., Meltzer P., Trent J..M. (1999) Expression profiling using cDNA microarrays, Nat. Genet. 21, 10–14.
- Dure L. (1993) A repeating 11-mer amino acid motif and plant desiccation, Plant J. 3, 363–369.
- Farooq M., Wahid A., Kobayashi N., Fujita D., Basra S.M.A. (2009) Plant drought stress: effects, mechanisms and management, Agron. Sustain. Dev. 29, 185–212.
- Fernie A.R., Geigenberger P., Stitt M. (2005) Flux an important, but neglected, component of functional genomics, Curr. Opin. Plant Biol. 8, 174–182.
- Fitter A.H., Hay R.K.M. (1981) Environmental Physiology of Plants, Academic Press, New York.
- Fowler S., Thomashow M.F. (2002) Arabidopsis transcriptome profiling indicated that multiple regulatory pathways are activated during cold acclimation in addition to the CBF cold response pathway, Plant Cell 14, 1675–1690.
- Frankow-Lindberg B.E. (2001) Adaptation to winter stress in nine white clover populations: changes in non-structural carbohydrates during exposure to simulated winter conditions and 'spring' regrowth potential, Ann. Bot. 88, 745–751.
- Ge L.F., Chao D.Y., Shi M., Zhu M.Z., Gao J.P., Lin H.X. (2008) Overexpression of the trehalose-6-phosphate phosphatase gene OsTPP1 confers stress tolerance in rice and results in the activation of stress responsive genes, Planta 228, 191–201.
- Gibbings J.G., Cook B.P., Dufault M.R., Madden S.L., Khuri S., Turnbull C.J., Dunwell J.M. (2003) Global transcript analysis of rice leaf and seed using SAGE technology, Plant Biotechnol. J. 1, 271–285.
- Gibson S., Arondel V., Iba K., Somerville C. (1994) Cloning of a temperature-regulated gene encoding a chloroplast

Omega-3 desaturase from *Arabidopsis thaliana*, Plant Physiol. 106, 1615–1621.

- Guy C.L. (1990) Cold acclimation and freezing stress tolerance: role of protein metabolism, Annu. Rev. Plant Phys. 41, 187–223.
- Hannah M.A., Wiese D., Freund S., Fiehn O., Heyer A.G., Hincha D.K. (2006) Natural genetic variation of freezing tolerance in *Arabidopsis*, Plant Physiol. 142, 98–112.
- Hansen J., Beck E. (1994) Seasonal changes in the utilization and turnover of assimilation products in a 8-year-old Scots pine (*Pinus sylvestris* L.) trees, Trees 8, 172–182.
- Hansen J., Türk R., Vogg G., Heim R., Beck E. (1997) Conifer carbohydrate physiology: Updating classical views, in: Rennenberg H., Eschrich W., Ziegler H. (Eds.), Trees – Contributions to Modern Tree Physiology, Leiden: Backhuys Publisher, pp. 97–108.
- Hernández-Nistal J., Dopico B., Labrador, E. (2002) Cold and salt stress regulates the expression and activity of a chickpea cytosolic Cu/Zn superoxide dismutase, Plant Sci. 163, 507–514.
- Hopkins W.G. (1999) The physiology of plants under stress, in: Introduction to Plant Physiology, 2nd ed., Wiley, New York, pp. 451–475.
- Houde M., Dhindsa R.S., Sarhan F. (1992) A molecular marker to select for freezing tolerance in Gramineae, Mol. Gen. Genet. 234, 43–48.
- Hsieh T.H., Lee J.T., Yang P.T., Chiu L.H., Charng Y.Y., Wang Y.C., Chan M.T. (2002) Heterology expression of the Arabidopsis C-repeat/dehydration response element binding factor 1 gene confers elevated tolerance to chilling and oxidative stresses in transgenic tomato, Plant Physiol. 129, 1086–1094.
- Hu H., You J., Fang Y., Zhu X., Qi Z., Xiong L. (2008) Characterization of transcription factor gene SNAC2 conferring cold and salt tolerance in rice, Plant Mol. Biol. 67, 169–181.
- Ingram J., Bartels D. (1996) The molecular basis of dehydration tolerance in plants, Annu. Rev. Plant Phys. 47, 377–403.
- Ito Y., Katsura K., Maruyama K., Taji T., Kobayashi M., Seki M., Shinozaki K., Yamaguchi-Shinozaki K. (2006) Functional analysis of rice DREB1/CBF-type transcription factors involved in cold-responsive gene expression in transgenic rice, Plant Cell Physiol. 47, 141–153.
- Jaglo K.R., Kleff S., Amundsen K.L., Zhang X., Haake V., Zhang J.Z., Deits T., Thomashow M.F. (2001) Components of the *Arabidopsis* C-repeat/dehydration responsive element binding factor cold-response pathway are conserved in *Brassica napus* and other plant species, Plant Physiol. 127, 910–917.
- Jaglo-Ottosen K.R., Gilmour S.J., Zarka D.G., Schabenberger O., Thomashow M.F. (1998) Arabidopsis CBF1 overexpression induces COR genes and enhances freezing tolerance, Science 280, 104–106.
- Jiang Q.W., Kiyoharu O., Ryozo I. (2002) Two novel mitogenactivated protein signaling components, OsMEK1 and OsMAP1, are involved in a moderate low-temperature signaling pathway in rice, Plant Physiol. 129, 1880–1891.
- Jones S.J., Riddle D.L., Pouzyrev A.T., Velculescu V.E., Hillier L., Eddy S.R., Stricklin S.L., Baillie D.L., Waterston R., Marra M.A. (2001) Changes in gene expression associated with developmental arrest and longevity in *Caenorhabditis elegans*, Genome Res. 11, 1346–1352.

- Kanneganti V., Gupta A.K. (2008) Overexpression of OsiSAP8, a member of stress associated protein (SAP) gene family of rice confers tolerance to salt, drought and cold stress in transgenic tobacco and rice, Plant Mol. Biol. 66, 445–462.
- Kaplan F., Kopka J., Haskell D.W., Zhao W., Schiller K.C., Gatzke N., Sung D.Y., Guy C.L. (2004) Exploring the temperature-stress metabolome of *Arabidopsis*, Plant Physiol. 136, 4159–4168.
- Kasuga M., Liu Q., Miura S., Yamaguchi-Shinozaki K., Shinozaki K. (1999) Improving plant drought, salt, and freezing tolerance by gene transfer of a single stress inducible transcription factor, Nat. Biotechnol. 17, 287–291.
- Kasuga M., Miura S., Shinozaki K., Yamaguchi-Shinozaki K. (2004) A combination of the *Arabidopsis* DREB1A gene and stress-inducible RD29A promoter improved drought and low-temperature stress tolerance in tobacco by gene transfer, Plant Cell Physiol. 45, 346–350.
- Katagiri T., Takahashi S., Shinozaki K. (2001) Involvement of a novel *Arabidopsis* phospholipase D, AtPLDd, in dehydration-inducible accumulation of phosphatidic acid in stress signaling, Plant J. 26, 595–605.
- Kim S., An C.S., Hong Y.N., Lee K.W. (2004) Cold-inducible transcription factor, CaCBF, is associated with a homeodomain leucine zipper protein in hot pepper (*Capsicum annuum* L.), Mol. Cell 18, 300–308.
- Komatsu S., Yang G., Khan M., Onodera H., Toki S., Yamaguchi M. (2007) Overexpression of calcium-dependent protein kinase 13 and calreticulin interacting protein 1 confers cold tolerance on rice plants, Mol. Genet. Genomics 277, 713–723.
- Kotak S., Larkindale J., Lee U., von Koskull-Doring P., Vierling E., Scharf K.D. (2007) Complexity of the heat stress response in plants, Curr. Opin. Plant Biol. 10, 310–316.
- Kovtun Y., Chiu W.L., Tena G., Sheen J. (2000) Functional analysis of oxidative stress activated mitogen-activated protein kinase cascade in plants, Proc. Natl Acad. Sci. (USA) 97, 2940–2945.
- Kranz H., Scholz K., Weisshaar B. (2000) c-MYB oncogene-like genes encoding three MYB repeats occur in all major plant lineages, Plant J. 21, 231–235.
- Kreps J.A., Wu Y., Chang H.S., Zhu T., Wang X., Harper J.F. (2002) Transcriptome changes for *Arabidopsis* in response to salt, osmotic, and cold stress, Plant Physiol. 130, 2129–2141.
- Krishna P., Sacco M., Cherutti J.F., Hill S. (1995) Cold-induced accumulation of hsp 90 transcripts in *Brasscia napus*, Plant Physiol. 107, 915–923.
- Kubien D.S., von Caemmerer S., Furbank R.T., Sage R.F. (2003) C4 photosynthesis at low temperature. A study using transgenic plants with reduced amounts of rubisco, Plant Physiol. 132, 1577–1585.
- Laloi C., Apel K., Danon A. (2004) Reactive oxygen signalling: the latest news, Curr. Opin. Plant Biol. 7, 323–328.
- Larkindale J., Hall J.D., Knight M.R., Vierling E. (2005) Heat stress phenotypes of *Arabidopsis* mutants implicate multiple signaling pathways in the acquisition of thermotolerance, Plant Physiol. 138, 882–888.
- Lee B.H., Henderson D.A., Zhu J.K. (2005) The Arabidopsis cold-responsive transcriptome and its regulation by ICE1, Plant Cell 17, 3155–3175.

- Lee B.H., Lee H., Xiong L., Zhu J.K. (2002) A mitochondrial complex I defect impairs cold-regulated nuclear gene expression, Plant Cell 14, 1235–1251.
- Levitt J. (1972) Responses of Plants to Environmental Stresses, Academic Press, New York.
- Li W., Li M., Zhang W., Welti R., Wang X. (2004) The plasma membrane-bound phospholipase Dd enhances freezing tolerance in *Arabidopsis thaliana*, Nat. Biotechnol. 22, 427–433.
- Liu K., Wang L., Xu Y., Chen N., Ma Q., Li F., Chong K. (2007) Overexpression of *OsCOIN*, a putative cold inducible zinc finger protein, increased tolerance to chilling, salt and drought, and enhanced proline level in rice, Planta 226, 1007–1016.
- Liu Q., Kasuga M., Sakuma Y., Abe H., Miura S., Yamaguchi-Shinozaki K., Shinozaki K. (1998) Two transcription factors, DREBI and DREB2, with an EREBP/AP2 DNA binding domain separate two cellular signal transduction pathways in drought and low temperature responsive gene expression respectively, in *Arabidopsis*, Plant Cell 10, 1391–1406.
- Llorente F., Oliveros J.C., Martinez-Zapater J.M., Salinas J. (2000) A freezing-sensitive mutant of *Arabidopsis*, *frs1*, is a new aba3 allele, Planta 211, 648–655.
- Lorenz W.W., Dean J.F. (2002) SAGE profiling and demonstration of differential gene expression along the axial developmental gradient of lignifying xylem in loblolly pine (*Pinus taeda*), Tree Physiol. 22, 301–310.
- Lynch D.V. (1990) Chilling injury in plants: the relevance of membrane lipids, in: Katterman F. (Ed.), Environmental Injury to Plants, Academic Press, New York, pp. 17–34.
- Mahajan S., Tuteja N. (2005) Cold, salinity and drought stresses: an overview, Arch. Biochem. Biophys. 444, 139–158.
- Maruyama K., Sakuma Y., Kasuga M., Ito Y., Seki M., Goda H., Shimada Y., Yoshida S., Shinozaki K., Yamaguchi-Shinozaki K. (2004) Identification of cold inducible downstream genes of the *Arabidopsis* DREB1A/CBF3 transcriptional factor using two microarray systems, Plant J. 38, 982–993.
- McKersie B.D., Bowley S.R. (1997) Active oxygen and freezing tolerance in transgenic plants, in: Plant Cold Hardiness, Molecular Biology, Biochemistry and Physiology, Plenum, New York, pp. 203–214.
- Meijer H.J., Munnik T. (2003) Phospholipid-based signaling in plants, Annu. Rev. Plant Biol. 54, 265–306.
- Meissner R.C., Jin H., Cominelli E., Denekamp M., Fuertes A., Greco R., Kranz H.D., Penfield S., Petroni K., Urzainqui A., Martin C., Paz-Ares J., Smeekens S., Tonelli C., Weisshaar B., Baumann E., Klimyuk V., Marillonnet S., Patel K., Speulman E., Tissier A.F., Bouchez D., Jones J.J., Pereira A., Wisman E., Bevan M. (1999) Function search in a large transcription factor gene family in *Arabidopsis*: assessing the potential of reverse genetics to identify insertional mutations in R2R3 MYB genes, Plant Cell 11, 1827–1840.
- Monroy A.F., Castonguay Y., Laberge S., Sarhan F., Vezina L.P., Dhindsa R.S. (1993) A new cold-induced alfalfa gene is associated with enhanced hardening at sub zero temperature, Plant Physiol. 102, 873–879.
- Morsy M.R., Almutairi A.M., Gibbons J., Yun S.J., de Los Reyes B.G. (2005) The OsLti6 genes encoding low-molecularweight membrane proteins are differentially expressed in rice cultivars with contrasting sensitivity to low temperature, Gene 344, 171–180.

- Mukhopadhyay A., Vij S., Tyagi A.K. (2004) Overexpression of a zinc-finger protein gene from rice confers tolerance to cold, dehydration, and salt stress in transgenic tobacco, Proc. Natl Acad. Sci. (USA) 101, 6309–6314.
- Munnik T. (2001) Phosphatidic acid: an emerging plant lipid second messenger, Trends Plant Sci. 6, 227–233.
- Nakashima K., Tran L.S., Van Nguyen D., Fujita M., Maruyama K., Todaka D., Ito Y., Hayashi N., Shinozaki K., Yamaguchi-Shinozaki K. (2007) Functional analysis of a NAC-type transcription factor OsNAC6 involved in abiotic and biotic stress responsive gene expression in rice, Plant J. 51, 617–630.
- Nayyar H., Chander K., Kumar S., Bains T. (2005) Glycine betaine mitigates cold stress damage in Chickpea, Agron. Sustain. Dev. 25, 381–388.
- Oh S.J., Song S.I., Kim Y.S., Jang H.J., Kim S.Y., Kim M., Kim Y.K., Nahm B.H., Kim J.K. (2005) Arabidopsis CBF3/DREB1A and ABF3 in transgenic rice increased tolerance to abiotic stress without stunting growth, Plant Physiol. 138, 341–351.
- Ohnishi T., Sugahara S., Yamada T., Kikuchi K., Yoshiba Y., Hirano H.Y., Tsutsumi N. (2005) OsNAC6, a member of the NAC gene family, is induced by various stresses in rice, Genes Genet. Syst. 80, 135–139.
- Olien C.R., Smith M.N. (1997) Ice adhesions in relation to freeze stress, Plant Physiol. 60, 499–503.
- Oono Y., Seki M., Nanjo T., Narusaka M., Fujita M., Satoh R., Satou M., Sakurai T., Ishida J., Akiyama K., Iida K., Maruyama K., Satoh S., Yamaguchi-Shinozaki K., Shinozaki K. (2003) Monitoring expression profiles of *Arabidopsis* gene expression during rehydration process after dehydration using ca. 7000 full-length cDNA microarray, Plant J. 34, 868–887.
- Orvar B.L., Sangwan V., Omann F., Dhindsa R.S. (2000) Early steps in cold sensing by plant cells: the role of actin cytoskeleton and membrane fluidity, Plant J. 23, 785–794.
- Pasquali G., Biricolti S., Locatelli F., Baldoni E., Mattana M. (2008) Osmyb4 expression improves adaptive responses to drought and cold stress in transgenic apples, Plant Cell Rep. 27, 1677–1686.
- Patankar S., Munasinghe A., Shoaibi A., Cummings L.M., Wirth D.F. (2001) Serial analysis of gene expression in *Plasmodium falciparum* reveals the global expression profile of erythrocytic stages and the presence of anti-sense transcripts in the malarial parasite, Mol. Biol. Cell 12, 3114–3125.
- Pellegrineschi A., Reynolds M., Pacheco M., Brito R.M., Almeraya R., Yamaguchi-Shinozaki K., Hoisington D. (2004) Stress-induced expression in wheat of the *Arabidopsis thaliana* DREB1A gene delays water stress symptoms under greenhouse conditions, Genome 47, 493–500.
- Pramanik M.H., Imai R. (2005) Functional identification of a trehalose 6-phosphate phosphatase gene that is involved in transient induction of trehalose biosynthesis during chilling stress in rice, Plant Mol. Biol. 58, 751–762.
- Qin F., Sakuma Y., Li J., Liu Q., Li Y-Q., Shinozaki K., Yamaguchi-Shinozaki K. (2004) Cloning and functional analysis of a novel DREB1/CBF transcription factor involved in cold-responsive gene expression in *Zea mays* L, Plant Cell Physiol. 45, 1042–1052.

- Quinn P.J. (1985) A lipid phase separation model of low temperature damage to biological membranes, Cryobiology 22, 28–46.
- Rabbani M.A., Maruyama K., Abe H., Khan M.A., Katsura K., Ito Y., Yoshiwara K., Seki M., Shinozaki K., Yamaguchi-Shinozaki K. (2003) Monitoring expression profiles of rice genes under cold, drought, and high-salinity stresses and abscisic acid application using cDNA microarray and RNA gelblot analyses, Plant Physiol. 133, 1755–1767.
- Rosinski J.A., Atchley W.R. (1998) Molecular evolution of the Myb family of transcription factors: evidence for polyphyletic origin, J. Mol. Evol. 46, 74–83.
- Saijo Y., Hata S., Kyozuka J., Shimamoto K., Izui K. (2000) Over-expression of a single Ca<sup>2+</sup>-dependent protein kinase confers both cold and salt/drought tolerance on rice plants, Plant J. 23, 319–327.
- Saijo Y., Kinoshita N., Ishiyama K., Hata S., Kyozuka J., Hayakawa T., Nakamura T., Shimamoto K., Yamaya T., Izui K. (2001) A Ca(2+)-dependent protein kinase that endows rice plants with cold- and salt-stress tolerance functions in vascular bundles, Plant Cell Physiol. 42, 1228–1233.
- Salomoni P., Perrotti D., Martinez R., Franceschi C., Calabretta B. (1997) Resistance to apoptosis in CTLL-2 cells constitutively expressing c-Myb is associated with induction of BCL-2 expression and Myb-dependent regulation of bcl-2 promoter activity, Proc. Natl Acad. Sci. (USA) 94, 3296–3301.
- Sangwan V., Foulds I., Singh J., Dhindsa R.S. (2001) Coldactivation of *Brassica napus* BN115 promoter is mediated by structural changes in membranes and cytoskeleton, and requires Ca<sup>2+</sup> influx, Plant J. 27, 1–12.
- Satoh R., Nakashima K., Seki M., Shinozaki K., Yamaguchi-Shinozaki K. (2002) ACTCAT, a novel cis-acting element for proline- and hypoosmolarity-responsive expression of the *ProDH* gene encoding proline dehydrogenase in *Arabidopsis*, Plant Physiol. 130, 709–719.
- Schwender J., Ohlrogge J., Shachar-Hill Y. (2004) Understanding flux in plant metabolic networks, Curr. Opin. Plant Biol. 7, 309–317.
- Seki M., Kamei A., Yamaguchi-Shinozaki K., Shinozaki K. (2003) Molecular responses to drought, salinity and frost: common and different paths for plant protection, Curr. Opin. Biotech. 14, 194–199.
- Seki M., Narusaka M., Abe H., Kasuga M., Yamaguchi-Shinozaki K., Carninci P., Hayashizaki Y., Shinozaki K. (2001) Monitoring the expression pattern of 1300 Arabidopsis genes under drought and cold stresses by using a fulllength cDNA microarray, Plant Cell 13, 61–72.
- Seki M., Narusaka M., Ishida J., Nanjo T., Fujita M., Oono Y., Kamiya A., Nakajima M., Enju A., Sakurai T., Satou M., Akiyama K., Taji T., Yamaguchi-Shinozaki K., Carninci P., Kawai J., Hayashizaki Y., Shinozaki K. (2002) Monitoring the expression profiles of 7000 Arabidopsis genes under drought, cold and high salinity stresses using a full-length cDNA microarray, Plant J. 31, 279–292.
- Seki M., Satou M., Sakurai T., Akiyama K., Iida K., Ishida J., Nakajima M., Enju A., Narusaka M., Fujita M., Oono Y., Kamei A., Yamaguchi-Shinozaki K., Shinozaki K. (2004) RIKEN Arabidopsis fullength (RAFL) cDNA and its applications for expression profiling under abiotic stress conditions, J. Exp. Bot. 55, 213–223.

- Senser M., Beck E. (1982) Frost resistance in spruce (*Picea abies* (L.) Karst): V. Influence of photoperiod and temperature on the membrane lipids of the needles, Z. Pflanzenphysiol. 108, 71–85.
- Shima S., Matsui H., Tahara S., Imai R. (2007) Biochemical characterization of rice trehalose-6-phosphate phosphatases supports distinctive functions of these plant enzymes, FEBS J. 274, 1192–1201.
- Shinozaki K., Yamaguchi-Shinozaki K. (2000) Molecular responses to dehydration and low temperature: differences and cross-talk between two stress signaling pathways, Curr. Opin. Plant Biol. 3, 217–223.
- Shinozaki K., Yamaguchi-Shinozaki K., Seki M. (2003) Regulatory network of gene expression in the drought and cold stress responses, Curr. Opin. Plant Biol. 6, 410–417.
- Somerville C. (1995) Direct tests of the role of membrane lipid composition in low temperature-induce photoinhibition and chilling sensitivity in plants and cyanobacteria, Proc. Natl Acad. Sci. (USA) 92, 6215–6218.
- Steponkus P.L. (1984) Role of the plasma membrane in freezing injury and cold acclimation, Ann. Rev. Plant Physiol. 35, 543–584.
- Steponkus P.L., Uemura M., Webb M.S. (1993) A contrast of the cryostability of the plasma membrane of winter rye and spring oat-two species that widely differ in their freezing tolerance and plasma membrane lipid composition, in: Steponkus P.L. (Ed.), Advances in Low-Temperature Biology, Vol. 2, JAI Press, London, pp. 211–312.
- Stockinge E.J., Gilmour S.J., Thomashow M.F. (1997) Arabidopsis thaliana CBF1 encodes an AP2 domain-containing transcription activator that binds to the C repeat/ DRE, a cisacting DNA regulatory element that stimulates transcription in response to low temperature and water deficit, Proc. Natl Acad. Sci. (USA) 94, 1035–1040.
- Stracke R., Werber M., Weisshaar B. (2001) The R2R3-MYB gene family in *Arabidopsis thaliana*, Curr. Opin. Plant Biol. 4, 447–456.
- Suzuki K., Nagasuga K., Okada M. (2008) The chilling injury induced by high root temperature in the leaves of rice seedlings, Plant Cell Physiol. 49, 433–442.
- Tao D.L., Öquist G., Wingsle G. G. (1998) Active oxygen scavengers during cold acclimation of scots pine seedlings in relation to freezing tolerance, Cryobiology 37, 38–45.
- Teige M., Scheikl E., Eulgem T., Doczi R., Ichimura K., Shinozaki K., Dangl J.L., Hirt H. (2004) The MKK2 pathway mediates cold and salt stress signaling in *Arabidopsis*, Mol. Cell 15, 141–152.
- Tester M., Bacic A. (2005) Abiotic stress tolerance in grasses. From model plants to crop plants, Plant Physiol. 137, 791–793.
- Testerink C., Munnik T. (2005) Phosphatidic acid: a multifunctional stress signaling lipid in plants, Trends Plant Sci. 10, 368–375.
- Thomashow M.F. (1999) Plant cold acclimation: freezing tolerance genes and regulatory mechanisms, Annu. Rev. Plant Phys. 50, 571–599.
- Tuteja R., Tuteja N. (2004) Serial analysis of gene expression (SAGE): unraveling the bioinformatics tools, Bioessays 26, 916–922.

- Uemura M., Steponkus P.L. (1997) Effect of cold acclimation on membrane lipid composition and freeze induced membrane destabilization, in: Plant Cold Hardiness, Molecular Biology, Biochemistry and Physiology, Plenum, New York, pp. 171–79.
- Vannini C., Locatelli F., Bracale M., Magnani E., Marsoni M., Osnato M., Mattana M., Baldoni E., Coraggio I. (2004) Overexpression of the rice Osmyb4 gene increases chilling and freezing tolerance of *Arabidopsis thaliana* plants, Plant J. 37, 115–127.
- Vaultier M.N., Cantrel C., Vergnolle C., Justin A.M., Demandre C., Benhassaine-Kesri G., Çiçek D., Zachowski A., Ruelland E. (2006) Desaturase mutants reveal that membrane rigidification acts as a cold perception mechanism upstream of the diacylglycerol kinase pathway in *Arabidopsis* cells, FEBS Lett. 580, 4218–4223.
- Velculescu V.E., Zhang L., Vogelstein B., Kinzler K.W. (1995) Serial analysis of gene expression, Science 270, 484–487.
- Velculescu V.E., Zhang L., Zhou W., Vogelstein J., Basrai M.A., Bassett D.E., Hieter P., Vogelstein B., Kinzler K.W. (1997) Characterization of the yeast transcriptome, Cell 88, 243–251.
- Vergnolle C., Vaultier M.N., Taconnat L., Renou J.P., Kader J.C., Zachowski A., Ruelland E. (2005) The cold-induced early activation of phospholipase C and D pathways determines the response of two distinct clusters of genes in *Arabidopsis* cell suspensions, Plant Physiol. 139, 1217–1233.
- Verslues P.E., Zhu J.K. (2005) Before and beyond ABA: upstream sensing and internal signals that determine ABA accumulation and response under abiotic stress, Biochem. Soc. T. 33, 375–379.
- Vinocur B., Altman A. (2005) Recent advances in engineering plant tolerance to abiotic stress: achievements and limitations, Curr. Opin. Biotechnol. 16, 123–132.
- Viswanathan C., Zhu J.K. (2002) Molecular genetic analysis of cold-regulated gene transcription, Philos. T. Roy. Soc. B 357, 877–886.
- Vogel J.T., Zarka D.G., Van Buskirk H.A., Fowler S.G., Thomashow M.F. (2005) Roles of the CBF2 and ZAT12 transcription factors in configuring the low temperature transcriptome of *Arabidopsis*, Plant J. 41, 195–211.
- Wan B., Lin Y., Mou T. (2007) Expression of rice Ca(2+)dependent protein kinases (CDPKs) genes under different environmental stresses, FEBS Lett. 581, 1179–1189.
- Wang L., Cai H., Bai X., Li L.W., Li Y., Zhu Y.M. (2008) Cultivation of transgenic rice plants with OsCDPK7 gene and its salt tolerance, Yi Chuan 30, 1051–1055.
- Wang Y.J., Zhang Z.G., He X.J., Zhou H.L., Wen Y.X., Dai J.X., Zhang J.S., Chen S.Y. (2003) A rice transcription factor OsbHLH1 is involved in cold stress response, Theor. Appl. Genet. 107, 1402–1409.
- Welti R., Li W., Li M., Sang Y., Biesiada H., Zhou H.E., Rajashekar C.B., Williams T.D., Wang X. (2002) Profiling membrane lipids in plant stress responses. Role of phospholipase Da in freezing induced lipid changes in *Arabidopsis*, J. Biol. Chem. 277, 31994–32002.

- Williams M.E., Torabinejad J., Cohick E., Parker K., Drake E.J., Thompson J.E., Hortter M., DeWald D.B. (2005) Mutations in the *Arabidopsis* phosphoinositide phosphatase gene SAC9 lead to overaccumulation of PtdIns(4,5)P<sub>2</sub> and constitutive expression of the stress-response pathway, Plant Physiol. 138, 686–700.
- Williams W.P. (1990) Cold-induced lipid phase transitions, Philos.T. Roy. Soc. B 326, 555–570.
- Xiang Y., Huang Y., Xiong L. (2007) Characterization of stressresponsive CIPK genes in rice for stress tolerance improvement, Plant Physiol. 144, 1416–1428.
- Xiong L., Yang Y. (2003) Disease resistance and abiotic stress tolerance in rice are inversely modulated by an abscisic acidinducible mitogen-activated protein kinase, Plant Cell 15, 745–759.
- Xiong L., Ishitani M., Lee H., Zhu J.K. (2001) The Arabidopsis LOS5/ABA3 locus encodes a molybdenum cofactor sulfurase and modulates cold stress- and osmotic stress-responsive gene expression, Plant Cell 13, 2063–2083.
- Yamaguchi-Shinozaki K., Shinozaki K. (1994) A novel cisacting element in an *Arabidopsis* gene is involved in responsiveness to drought, low temperature, or high-salt stress, Plant Cell 6, 251–264.
- Yamaguchi-Shinozaki K., Shinozaki K. (2005) Organization of cis-acting regulatory elements in osmotic- and coldstressresponsive promoters, Trends Plant Sci. 10, 88–94.
- Yamaguchi-Shinozaki K., Shinozaki K. (2006) Transcriptional regulatory networks in cellular responses and tolerance to dehydration and cold stresses, Annu. Rev. Plant Biol. 57, 781–803.
- Yanhui C., Xiaoyuan Y., Kun H., Meihua L., Jigang L., Zhaofeng G., Zhiqiang L., Yunfei Z., Xiaoxiao W., Xiaoming Q., Yunping S., Li Z., Xiaohui D., Jingchu L., Xing-Wang D., Zhangliang C., Hongya G., Li-Jia Q. (2006) The MYB transcription factor superfamily of *Arabidopsis*: expression analysis and phylogenetic comparison with the rice MYB family, Plant Mol. Biol. 60, 107–124.
- Zhang J.Z., Creelman R.A., Zhu J.K. (2004) From laboratory to field. Using information from *Arabidopsis* to engineer salt, cold, and drought tolerance in crops, Plant Physiol. 135, 615–621.
- Zhang W., Wang C., Qin C., Wood T., Olafsdottir G., Welti R., Wang X. (2003) The oleate-stimulated phospholipase D, PLDd, and phosphatidic acid decrease H<sub>2</sub>O<sub>2</sub>-induced cell death in *Arabidopsis*, Plant Cell 15, 2285–2295.
- Zhang W., Yu L., Zhang Y., Wang X. (2005) Phospholipase D in the signaling networks of plant response to abscisic acid and reactive oxygen species, Biochim. Biophys. Acta 1736, 1–9.
- Zhu J., Dong C.H., Zhu J.K. (2007) Interplay between coldresponsive gene regulation, metabolism and RNA processing during plant cold acclimation, Curr. Opin. Plant Biol. 10, 290–295.
- Zhu J., Verslues P.E., Zheng X., Lee B.H., Zhan X., Manabe Y., Sokolchik I., Zhu Y., Dong C.H., Zhu J.K., Hasegawa P.M., Bressan R.A. (2005) HOS10 encodes an R2R3-type MYB transcription factor essential for cold acclimation in plants, Proc. Natl Acad. Sci. (USA) 102, 9966–9971.

## Part V Alternative Pest Control

## Defence Mechanisms of Brassicaceae: Implications for Plant-Insect Interactions and Potential for Integrated Pest Management

Ishita Ahuja, Jens Rohloff, and Atle Magnar Bones

## Contents

1	Intro	luction	624
	1.1	The Origin and Excellence of "Brassicas"	624
	1.2	Important Insect Pests of Brassicas	625
	1.3	Plant Defence Mechanisms – General	
		Information	631
2	Defen	ce Components of Brassicaceae:	
	Gluco	sinolates, the Glucosinolate-Myrosinase	
	Syster	n, Plant Volatiles, Phytoalexins,	
	Phyto	anticipins and Sulphur	632
	2.1	Glucosinolates	632
	2.2	The Glucosinolate-Myrosinase Defence	
		System	633
	2.3	Plant Volatiles	633
	2.4	Phytoalexins and Phytoanticipins	635
	2.5	Sulphur	635
3	Chem	ical Ecology and Insect Behavioural Aspects	636
3	<b>Chem</b> 3.1	ical Ecology and Insect Behavioural Aspects . Oviposition	636 636
3	<b>Chem</b> 3.1 3.2	ical Ecology and Insect Behavioural Aspects . Oviposition Trophic Interactions Among <i>Brassica</i>	636 636
3	<b>Chem</b> 3.1 3.2	ical Ecology and Insect Behavioural Aspects . Oviposition Trophic Interactions Among <i>Brassica</i> Crops-Herbivores-Parasitoids-	636 636
3	<b>Chem</b> 3.1 3.2	ical Ecology and Insect Behavioural Aspects . Oviposition Trophic Interactions Among <i>Brassica</i> Crops-Herbivores-Parasitoids- Hyperparasitoids	636 636 644
3	Chem 3.1 3.2 Brass	ical Ecology and Insect Behavioural AspectsOvipositionTrophic Interactions Among BrassicaCrops-Herbivores-Parasitoids-Hyperparasitoidsicas and Their Potential for Integrated Pest	636 636 644
3	Chem 3.1 3.2 Brass Mana	ical Ecology and Insect Behavioural Aspects         Oviposition         Trophic Interactions Among Brassica         Crops-Herbivores-Parasitoids-         Hyperparasitoids         icas and Their Potential for Integrated Pest         gement	636 636 644 653
3	Chem 3.1 3.2 Brass Mana 4.1	ical Ecology and Insect Behavioural Aspects         Oviposition         Trophic Interactions Among Brassica         Crops-Herbivores-Parasitoids-         Hyperparasitoids         icas and Their Potential for Integrated Pest gement         Brassicas as Biocontrol Agents	636 636 644 653 654
3	Chem 3.1 3.2 Brass Mana 4.1 4.2	ical Ecology and Insect Behavioural Aspects         Oviposition         Trophic Interactions Among Brassica         Crops-Herbivores-Parasitoids-         Hyperparasitoids         icas and Their Potential for Integrated Pest gement         Brassicas as Biocontrol Agents         Brassicas as Cover Crops	636 636 644 653 654 655
3	Chem 3.1 3.2 Brassi Mana 4.1 4.2 4.3	ical Ecology and Insect Behavioural AspectsOvipositionTrophic Interactions Among BrassicaCrops-Herbivores-Parasitoids-Hyperparasitoidsdicas and Their Potential for Integrated PestgementBrassicas as Biocontrol AgentsBrassicas as Cover CropsBrassicas as Trap Crops	636 636 644 653 654 655 655
3	Chem 3.1 3.2 Brass: Mana 4.1 4.2 4.3 4.4	ical Ecology and Insect Behavioural Aspects         Oviposition         Trophic Interactions Among Brassica         Crops-Herbivores-Parasitoids-         Hyperparasitoids         icas and Their Potential for Integrated Pest gement         Brassicas as Biocontrol Agents         Brassicas as Cover Crops         Brassicas as Trap Crops         Plant Breeding Towards Insect Resistance	636 636 644 653 654 655 655
3	Chem 3.1 3.2 Brass Mana 4.1 4.2 4.3 4.4	ical Ecology and Insect Behavioural AspectsOvipositionTrophic Interactions Among BrassicaCrops-Herbivores-Parasitoids-HyperparasitoidsGras and Their Potential for Integrated PestgementBrassicas as Biocontrol AgentsBrassicas as Cover CropsBrassicas as Trap CropsPlant Breeding Towards Insect Resistancein Brassica Crops	636 636 644 653 654 655 655 655
3 4 5	Chem 3.1 3.2 Brassi Mana 4.1 4.2 4.3 4.4 Concl	ical Ecology and Insect Behavioural AspectsOvipositionTrophic Interactions Among BrassicaCrops-Herbivores-Parasitoids-HyperparasitoidsHyperparasitoidsBrassicas and Their Potential for Integrated PestgementBrassicas as Biocontrol AgentsBrassicas as Cover CropsBrassicas as Trap CropsPlant Breeding Towards Insect Resistancein Brassica Cropsusions and Future Directions	636 636 644 653 654 655 655 655

#### I. Ahuja (🖂)

Abstract Brassica crops are grown worldwide for oil, food and feed purposes, and constitute a significant economic value due to their nutritional, medicinal, bioindustrial, biocontrol and crop rotation properties. Insect pests cause enormous yield and economic losses in Brassica crop production every year, and are a threat to global agriculture. In order to overcome these insect pests, Brassica species themselves use multiple defence mechanisms, which can be constitutive, inducible, induced, direct or indirect depending upon the insect or the degree of insect attack. Firstly, we give an overview of different Brassica species with the main focus on cultivated brassicas. Secondly, we describe insect pests that attack brassicas. Thirdly, we address multiple defence mechanisms, with the main focus on phytoalexins, sulphur, glucosinolates, the glucosinolate-myrosinase system and their breakdown products. In order to develop pest control strategies, it is important to study the chemical ecology, and insect behaviour. We review studies on oviposition regulation, multitrophic interactions involving feeding and host selection behaviour of parasitoids and predators of herbivores on brassicas. Regarding oviposition and trophic interactions, we outline insect oviposition behaviour, the importance of chemical stimulation, oviposition-deterring pheromones, glucosinolates, isothiocyanates, nitriles, and phytoalexins and their importance towards pest management. Finally, we review brassicas as cover and trap crops, and as biocontrol, biofumigant and biocidal agents against insects and pathogens. Again, we emphasise glucosinolates, their breakdown products, and plant volatile compounds as key components in these processes, which have been considered beneficial in

Department of Biology, Norwegian University of Science and Technology, Realfagbygget, NO-7491 Trondheim, Norway e-mail: atle.bones@bio.ntnu.no

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_28,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Ahuja et al., Agron. Sustain. Dev. 30 (2010) 311–348. DOI: 10.1051/agro:2009025

the past and hold great prospects for the future with respect to an integrated pest management.

Keywords Brassicas · Insect pests · Chemical ecology

- Trophic levels Glucosinolates Isothiocyanates
- Defence mechanisms Biocontrol Trap crops
- Integrated pest management

## 1 Introduction

## 1.1 The Origin and Excellence of "Brassicas"

"Brassicas: Oil-, food- and fodder-bearing crops with small seeds; that can grow as fast as Wisconsin rapid cycling brassicas; can grow as big as ornamental plants; can be seen as vast green fields of vegetable crops or as oilseed crops with fields of bright yellow flowers; lead to the production of economically important agricultural products; used as food for humans and animals; are important as valuable renewable bioenergy resources; are huge reservoirs of plant innate defences; show multiple defence responses in response to stresses; possess anticancer properties; hold ample potential for pest management" (Ishita Ahuja).

The Brassicaceae or Cruciferae, also known as crucifers, is a broad family of around 375 genera and 3200 species (LeCoz and Ducombs, 2006), which includes crops, ornamentals and many weeds. The genus *Brassica* belongs to the subtribe Brassicinae of the Brassicaceae family and comprises about 159 species, which, along with cultivated species, also includes wild brassicas (Zhou and Zhang, 2001; Zhang and Zhou, 2006).

The cultivated species of Brassicaceae include oilseed rape (Fig. 1), cabbage, cauliflower, broccoli, Brussels sprouts, turnip, kale, swede, various mustards and other leafy vegetables. They are grown worldwide with a wide spectrum of adaptation for cultivation under varied agro-climatic conditions (Suwabe et al., 2006; Hong et al., 2008). The archaeological evidence of importance of brassicas dates back to 5000 BC, and *Brassica* plants are considered among the oldest cultivated plants known to humans, with written records dating back to ca. 1500 BC (Raymer, 2002). The genetic relationship among different *Brassica* species was established in the classical work by U (U.N., 1935), which is now ascribed as U's triangle (Fig. 2).

The corners of the U-triangle include three diploid species: *B. rapa* L. (2n = 20; AA), *B. nigra* L. Koch (2n = 16; BB) and *B. oleracea* L. (2n = 18; CC), and the other three in the middle of the triangle are amphidiploid species: *B. napus* L. (2n = 38; AACC), *B. juncea* (L.) Czern. (2n = 36; AABB) and *B. carinata* Braun (2n = 34; BBCC) (Gomez-Campo, 1999). These six species of the U-triangle are also referred to as: *B. rapa* (*syn. B. campestris*) (Chinese cabbage and Turnip), *B. nigra* (Black mustard), *B. oleracea* (Cabbage, Brussels sprouts, Cauliflower and Broccoli),



Fig. 1 The classical view of yellow Brassica napus fields and a closer view of oilseed rape plants



**Fig. 2** U-triangle showing the affiliations among different *Brassica* species (U.N., 1935)

[B. napus var. oleifera (Oilseed rape, Rapeseed and Canola) var. rapifera (Swede, Rutabaga)], B. juncea (Indian mustard/Brown mustard) and B. carinata (Ethopian mustard) (Labana and Gupta, 1993). B. rapa, B. nigra, B. oleracea, B. juncea, B. napus and *B*. carinata are all designated as crop brassicas. Cauliflower, cabbage, Brussels sprouts, broccoli and turnip fall into the category of important vegetable crops. Among these vegetable crops, broccoli (var. botrytis) has been a valuable asset in the USA, both from a nutritional and economic point of view. Apart from being an excellent source of vitamin C, it provides dietary fibre, protein, iron, calcium and vitamin A and also contains anticancer components (Verhoeven et al., 1997; Cintas et al., 2002). Brassica crops carry properties of nutritional value, health benefits and biocontrol agents, and their use in crop rotations has been valued both by traditional and organic farmers (Guerena, 2006). The worldwide production of cabbages and kale-like brassicas in 2007 was 69 Mio t/3.1 Mioha area, rapeseed 50 Miot/30.2 Mio ha, cauliflowers and broccolis 19 Mio/1.0 Mio ha, and mustard seed 0.4 Mio t/0.7 Mio ha (FAOSTAT, 2009). Due to the increased oilseed production of *B*. rapa, B. juncea, B. napus and B. carinata in the past three decades, the oilseed brassicas have become an important source of oil and protein among cultivated brassicas (Font et al., 2003). Oilseed Brassica species are the major oilseed crops cultivated in India and around the world, and India produces about 11.3% of the world's rapeseed mustard (Damodaram and Hegde, 2002 as cited by Chattopadhyay et al., 2005). Oilseed Brassica species are also an important commodity in

the world economy, as they are an important source of nutrition in developing countries (Rana, 2005). Amidst oilseed brassicas, oilseed rape has become a major crop in Europe, and one of the significant oil crops worldwide (Graner et al., 2003; Dubuis et al., 2005). The rapeseed and rapeseed oil production in 2007 from different countries that ranked in the first twelve worldwide are listed in Table 1.

B. napus stands as the third most important oilseed crop at the international level for both oil meal (after soybean and cotton) and vegetable oil (after soybean and oil palm) (Snowdon et al., 2007). The worldwide rapeseed oil production was 16.8 Mio t after palm and soybean (FAOSTAT, 2009). The significance of oilseed rape has increased during the present decade, not only because of its consumption as a nutritional food, but also due to its role as a renewable energy source as biodiesel in transport. Biodiesel is the methyl ester of oilseed rape oil (RME) (Souckova, 2006). B. juncea, an important source of edible oil, is cultivated in many countries across the globe. During the year 1999–2000, oilseed rape and mustard occupied, e.g., 6 Mio ha with an annual production of 5.8 Miot in India (Dutta et al., 2005).

### 1.2 Important Insect Pests of Brassicas

"An insect is considered a pest if it threatens a resource valued by human beings, including human health" (Foster and Harris, 1997). Insect pests pose a great challenge to *Brassica* crop production worldwide. A huge number of insect pests attack brassicas, and several insect specialists have *Brassica* species as preferred host plants (Lamb, 1989; Sekhon and Åhman, 1993; Sibanda et al., 2000). These insect pests are the major un-equalisers of growth and crop yield of brassicas and their importance varies by geographical location (Hokkanen and Wearing, 1996; Kanrar et al., 2002). Some of the major insect pests that attack Brassicacea crops worldwide are listed in Table 2 and presented in Fig. 3.

Sources: AgroAtlas, 2009; Bromand, 1990; Bartlet et al., 1996; Hokkanen and Wearing, 1996; Ruther and Thiemann, 1997; Girard et al., 1998; Kift et al., 2000; Ulmer, 2002; Ester et al., 2003; Du Toit, 2007; Kazana et al., 2007; Khattab, 2007; Lehrman, 2007; Valantin-Morison et al., 2007; Smallegange et al., 2007; Knodel and Ganehiarachchi, 2008; Cartea et al., 2009. **Table 1** Rapeseed andrapeseed oil production inthe first 12 highest producingcountries worldwide in 2007(FAOSTAT, 2009)

Country	Rapeseed (Mio t)	Rapeseed oil (Mio t)
China	10.38	4.35
Canada	8.86	1.30
India	7.10	2.34
Germany	5.32	2.20
France	4.55	1.05
Poland	2.11	0.68
United Kingdom	2.11	0.72
Australia	1.07	0.17
Ukraine	1.06	0.06
Czech Republic	1.04	0.31
USA	0.66	0.50
Russian Federation	0.60	0.07

### 1.2.1 The Diamondback Moth (Plutella xylostella)

The diamondback moth (P. xylostella) (Fig. 3) is a highly mobile insect, and is considered as the most damaging insect pest of Brassica crops worldwide. With an estimated control cost of nearly US\$ 1 billion annually (Talekar and Shelton, 1993; Pivnick et al., 1994; Sarfraz et al., 2006; Golizadeh et al., 2007; Shelton et al., 2008), it has been the greatest threat to Brassica production in many regions of the world, with crop loss of 90% in some cases (Verkerk and Wright, 1996; Charleston and Kfir, 2000). The diamondback moth was first noticed as a pest in South Africa in the early 1900s (Charleston and Kfir, 2000). In Canada, the short generation time and high fecundity of the diamondback moth allows it to become a significant pest of oilseed crops in this region (Ulmer et al., 2002). Diamondback moth can attack plants at all stages of growth and female moths attach their eggs singly or in groups of two or three to the underside of leaves. Larval chewing make small holes in leaves, with larger larvae making larger holes. Their chewing may make leaves appear "windowpaned", with a clear cuticle left after feeding. Diamondback moth larvae feed on most of the *Brassica* plants such as *B*. *campestris*, *B*. *napus*, B. juncea and B. oleracea.

#### 1.2.2 The Cabbage Looper (Trichoplusia ni)

The cabbage looper feeds on a diverse range of plants and is an important pest on Brassicaceae plants (Chow et al., 2005). The cabbage loopers (Fig. 3), which often attack broccoli, cauliflower, cabbage, kale, collards and mustard, are pale green larvae of a grey-brown moth, feed on foliage and tender above-ground parts and generally lead to plant decline (Du Toit, 2007; Capinera, 2008). Young larvae initially are dusky white, but become pale green as they begin to feed on foliage. The cabbage looper is found throughout Canada, Mexico and the United States, wherever crucifers are cultivated, and on other continents (Capinera, 2008). The cabbage looper females can produce 300 to 600 eggs in their 2 week life span, laying eggs singly and mostly on the lower surface of leaves. Larvae emerge from eggs in 3 to 4 days and feed on leaves (Chow et al., 2005; Mossler, 2005).

### 1.2.3 The Cabbage Moth (*Mamestra brassicae*) and Bertha Armyworm (*Mamestra configrata*)

The cabbage moth, M. brassicae (Fig. 3), is a polyphagous insect, and the observed food plants of the cabbage moth include more than 70 species of 22 families, of which Brassicaceae and Chenopodiaceae are among the most preferred (Popova, 1993 as cited by Rojas et al., 2001; Ulland et al., 2008). The cabbage moth is widely distributed throughout most of Europe and Asia, from 30°N to about 70°N (Klingen et al., 2002b and references therein). The cabbage moth is native to Norway and is an important pest on various cabbage crops in Southern Norway up to about 62°N (Johansen, 1997). The Bertha armyworm, also referred to as the 'Miller Moth' or 'climbing cutworm', is native to North America and is a pest of oilseed and canola production in the northern Great Plains (Ulmer, 2002; Knodel and Ganehiarachchi, 2008).

Tabl	e 2 List of some important insect pests that attack Bra	ssicaceae plants worldwide	
COH C	imon names	Scientific names	Host plants
Lepi	doptera		
	Diamondback moth/Cabbage moth (Specialist)	Plutella xylostella L.	Broccoli, Brussels sprouts, cabbage, cauliflower, kale, mustard, turnin
5	Cabbage looper (Generalist)	Trichoplusia ni Hübner	Broccoli, cabbage, cauliflower, kale, collards, mustard, michoros murito
З	Cabbage moth (Generalist)	Mamestra brassicae L.	Cabbage, mustard, turnip,
4	Bertha armyworm (Generalist)	Mamestra configrata Walker	Canola, rapeseed, mustard
2	Small white butterfly/Imported cabbage worm/ Cabbage butterfly (Specialist)	Pieris rapae L.	Broccoli, Brussels sprouts, cabbage, cauliflower, kale,
9	Large white butterfly (Specialist)	Pieris brassicae L.	Kale, cabbage, turnip, black mustard, Ethiopian mustard, survede
7	Green-veined white (Specialist)	Pieris napi L.	Cabbage, Brussels sprouts, turnip, swede, mustard, oilseed rape
Hyn	tenoptera		
8	Cabbage sawfly/Turnip sawfly (Specialist)	Athalia rosae L.	Mustard, turnip, oilseed rape, cabbage
Dipt	era		
6	Cabbage maggot/Cabbage root fly/Cabbage fly (Specialist)	Delia radicum syn. brassicae L.	Broccoli, Brussels sprouts, cabbage, kale, swede
10	Turnip root fly (Specialist)	Delia floralis Fallen	Turnip
11 Cole	Brassica/Pod midge (Specialist)	Dasineura brassicae Winnertz.	Oilseed rape
12	Flea beetles/pollen beetles Crucifer flea beetle (Specialist)	Phyllotreta, Psylliodes and Meligethes spp. Phyllotreta cruciferae Goeze	Broccoli, Brussels sprouts, cabbage, cauliflower, kale,
	Striped flea beetle (Specialist)	Phyllotreta striolata F.	Broccoli, Brussels sprouts, cabbage, cauliflower, kale, collards, canola, turnip, mustard, oilseed rape
13	Cabbage stem flea beetle (Specialist) Pollen beetles Weevils	Psylliodes chrysophela L. Meligethes spp. Ceutorhynchus spp.	Cabbage, mustard, turnip, oilseed rape Turnip, swede, mustard, oilseed rape
2	Rane stem weevil (Snecialist)	C nami Gull	Oilseed rane cabhage
	Cabbage seed weevil (Specialist)	C. assimilis Pavkull	Cabbage, turnip, oilseed rape
	Cabbage seedpod weevil (Specialist)	C. obstrictus Marsham	Turnip, mustard, cabbage, brown mustard, canola,
Hom	Cabbage stem weevil (Specialist) notera	C. pallidactylus Marsham	Concepts, proceed applies of the calliflower, turnip
14	Aphids		
	Cabbage aphid/Mealy cabbage aphid (Specialist)	Brevicoryne brassicae L.	Cabbage, mustard, turnip, oilseed rape, broccoli, Brussels
	Mustard aphid (Specialist)	Lipaphis erysimi Kaltenbach	Mustard
	Turnip aphid (Specialist)	Hyadaphis erysimi Kaltenbach	Turnip
	Green peach aphid (Generalist)	Myzus persicae Sulzer	Broccoli, Brussels sprouts, cabbage, cauliflower, turnip



Fig. 3 Important insect pests that attack Brassicaceae plants worldwide

Young larvae feed on the underside of leaves and chewing makes irregular-shaped holes. The economic damage occurs due to significant larval feeding on foliage as well as on developing seedpods of canola. In years with outbreaks, larval feeding has resulted in economic crop losses and increased production costs from spraying insecticides (Knodel and Ganehiarachchi, 2008).

# 1.2.4 The Cabbage White Butterflies (Pieris brassicae, Pieris rapae and Pieris napi)

The cabbage white butterflies P. brassicae, P. rapae and P. napi (Fig. 3), are specialised on the Brassicaceae family and they have been used as a model species in the field of insect pest biology (Smallegange et al., 2007 and references therein). P. brassicae and P. rapae are cabbage herbivores that are closely related, yet show drastic contrasts in the amount of eggs they lay on plants (Bruinsma et al., 2007). P. rapae occurs in temperate regions around the world, and is generally confused with other common cabbage white butterflies (Capinera, 2004). In North America, it is known as "imported cabbageworm" and in Europe it is known as the small white cabbage butterfly. P. rapae is a cosmopolitan species, which is widespread throughout Europe, Asia and North America. The damage to foliage caused by P. rapae is slight, although it can be severe in seasons with a high infestation of caterpillars (Hern et al., 1996). P. napi is spread throughout the northern hemisphere ranging from North America, Europe and Asia to North Africa. Although larvae mainly feed on wild Brassicaceae species, infestation of Brassica vegetable crops such as cabbage, turnip and swede occurs and potentially leads to significant crop losses.

### 1.2.5 The Cabbage Sawfly/Turnip Sawfly (Athalia rosae)

The turnip sawfly, *A. rosae* (Fig. 3), is oligophagous, i.e. it feeds on a few types of plants in nature. The pest eats leaf mass, buds, flowers and young pods (AgroAtlas, 2009). Among cruciferous plants its preferred hosts are white mustard (*S. alba*) and turnip (*B. rapa*), but the pest can also rear on young oilseed rape (*B. napus*) crops (Barker et al., 2006).

## 1.2.6 The Cabbage Root Fly (*Delia radicum* syn., brassicae) and the Turnip Root Fly (*Delia florales*)

The cabbage root fly (D. radicum syn. brassicae) (Fig. 3) and the turnip root fly (D. florales) are considered to be economically important pests on Brassica crops such as broccoli, Brussels sprouts, cabbages and kales, and are members of a large family of root flies (Klingen et al., 2002b and references therein; Ester et al., 2003). The larvae of Delia flies cause damage to plants by feeding on plant roots and through eggs and neonates (Klingen et al., 2002b). The plant mortality rate is unusually high, and the recovered plants are of reduced marketable quality (De Jong and Städler, 1999). Cabbage root fly and turnip root fly damaging Brassica spp. roots lead to significant reductions in yield, flowering and seed production as well as leaf, stem and root biomass (Blossey and Hunt-Joshi, 2003). Both cabbage root fly and turnip root fly are oligophagous in nature. Gravid females of the cabbage root fly arrive at a host-plant odour source by a series of short upwind flights, landing and reorientating into the wind between flights (Hopkins et al., 1999). The adult female oviposits in the soil close to the stem of Brassica plants (Nottingham and Coaker, 1985; De Jong and Städler, 1999), while damage to the plants is caused by hatched larvae feeding on the roots.

### 1.2.7 The Brassica Pod Midge (Dasineura brassicae)

The brassica pod midge is a common pest on Brassicaceae plants, particularly on oilseed rape, throughout north-western Europe and is attacked by over 20 species of hymenopteran parasitoids, among them *Omphale clypealis* and members of the genus *Platygaster subuliformis* (Murchie et al., 1997 and references therein; Murchie and Hume, 2003). Females deposit their eggs inside *Brassica* pods, where larvae feed and develop. Larvae spend their feeding period within the same pod (Åhman, 1985).

# 1.2.8 The Flea and Pollen Beetles (*Phyllotreta, Psylliodes* and *Meligethes* spp.)

*Phyllotreta nemorum* and *Phyllotreta undulata* are the main flea beetle species infesting *Brassica* crops, such as Brussels sprouts, broccoli and cauliflower (Ester et al., 2003). The crucifer flea beetle, *Phyllotreta cruciferae*, and the striped flea beetle, *Phyllotreta striolata*, are the most serious insect pests of canola in North America and both were introduced from Eurasia. *P. cruciferae* has become the dominant flea beetle pest of canola. Adult flea beetles emerge in the spring and

feed on the cotyledons and true leaves. When emerging in huge numbers, they can quickly devastate a seedling canola field. Therefore, a timely detection and management of this pest is important. The damage to oilseed Brassica crops through flea beetles exceeds US \$ 300 million annually in North America (Knodel and Olsen, 2002). Psylliodes chrysocephala (cabbage stem flea beetle) is the major stem-mining pest of oilseed rape (Barari et al., 2005) and an important pest of other Brassica spp. in Europe (Bromand, 1990; Bartlet and Williams, 1991; Bartlet et al., 1996; Vig, 2002; Valantin-Morison et al., 2007). The genus Meligethes, generally known as pollen beetles (Fig. 3), occurs worldwide and includes more than 400 species (Kirk-Spriggs, 1996 as cited by Blight and Smart, 1999). Of the 10 Meligethes species reported from brassicaceous plants in Europe, M. aeneus is by far the most common on cultivated brassicas (Blight and Smart, 1999). Pollen beetles are considered important pests on oilseed brassicas and cause serious yield losses to oilseed rape crops, with yield reductions of more than 80% reported (Lamb, 1989; Ekbom, 1995; Ekbom and Borg, 1996; Ruther and Thiemann, 1997; Ekbom and Ferdinand, 2003; Bartlet et al., 2004; Hansen, 2004; Williams, 2006; Kazachkova, 2007; Lehrman, 2007). Although pollen beetles are important pests, particularly for oilseed rape in Northern Europe (Jönsson and Anderson, 2007), and turnip rape, they do not oviposit on all Brassicacea plants (Bartlet et al., 2004).

#### 1.2.9 The Weevils (Ceutorhynchus spp.)

The cabbage seed weevil (C. assimilis) (Fig. 3) is an important pest of oilseed rape in Europe and North America (Lamb, 1989; Bartlet et al., 1997; Smart and Blight, 1997; Girard et al., 1998; Ferguson et al., 1999b), causing reduction in crop yield in heavily infested fields (Smart et al., 1997; Girard et al., 1998). The cabbage seed weevil is an oligophagous insect that feeds and develops on Brassica species. The cabbage seed weevil invades oilseed rape crops, the adults colonise flowering host plants and feed on pollen, and the larvae feed on the developing seeds before leaving the pods and pupating in the soil. The cabbage seedpod weevil (C. obstrictus) is native to Europe, and is a serious pest of brassicaceous oilseed crops in Europe and North America (Bartlet et al., 1993; Ferguson et al., 1999b; Ulmer and Dosdall,

2006; Valantin-Morison et al., 2007). The cabbage seedpod weevil is a small, dark grey "snout beetle" normally occurring after peak flowering and lays eggs in the pods of cruciferous plants. The larvae feed on seeds in the pods, which results in seed loss (Bartlet et al., 1993; Du Toit, 2007). When mature, the larvae leave the pods, and fall on the ground to pupate. Adults emerge in late July and feed before diapausing until the following spring. The cabbage seedpod weevil therefore needs to find a host plant at two stages of its life cycle: upon emergence from pupation in the summer (pre-diapause weevils), and upon emergence from hibernation the following spring (post-diapause weevils) (Bartlet et al., 1993). The cabbage stem weevil (C. pallidactylus) is the major stem-mining pest of oilseed rape (Barari et al., 2005).

# 1.2.10 The Aphids (Brevicoryne brassicae, Myzus persicae and Lipaphis erysimi)

Aphids have been considered as the main insect pests in India and northern European agriculture, and are also important pests in horticulture both in field and greenhouse production, causing damage to crops either directly by feeding or by transmitting plant viral diseases (Dawson et al., 1990; Sekhon, 1999). Aphids feed by piercing plant tissue with their needle-like mouthparts (stylets), sucking water and nutrients from the phloem vascular system of the plant. Feeding damage and toxins in the saliva cause thickening, crumpling and downward curling of leaves (Mossler, 2005). The nymphs and adults suck sap from leaves, stem, flowers and pods, hence resulting in poor pod formation and reduced oil content in grains. A severe aphid attack can result in up to 75% loss to Brassica crops (Sekhon, 1999). B. brassicae (Fig. 3) is a global problem with a strong negative impact on agriculture and horticulture. B. brassicae is controlled by multiple insecticide treatments (Kift et al., 2000; Pontoppidan et al., 2003) and is a severe pest on brassicas (Cole, 1994; Kift et al., 2000). B. brassicae is highly host-specific, feeding almost exclusively on the phloem sap of Brassica or other closely related plant species (Cole, 1997). B. brassicae produces parthenogenic, viviparous females throughout the year, which overwinter on horticultural brassicas and forage crops of rape and swede (Schroeder and Dumbleton, 2001). During spring, these females

change into winged forms, fly to seedlings of brassicas and produce offspring (nymphs). The peach aphid (M. persicae) is a generalist (Fig. 3) reported to have more than 400 species as host plants (Quaglia et al., 1993 as cited by Francis et al., 2001). The mustard aphid (L. erysimi) is the most important pest of cruciferous crops worldwide, causing damage of 10-90% depending upon the severity of the infestation and crop stage. Apart from causing damage as a sapsucker, it is also a vector of several viral diseases (Rana, 2005). A 2-year study on the preference and performance of L. erysimi on different Brassica species in the field and under greenhouse conditions revealed that rapeseed (B. campestris varieties BSH-1 and YSPB-9) and mustard (B. juncea RH-30) were better hosts for this aphid than other Brassica species (B. napus, B. nigra, B. carinata) (Rana, 2005). Moreover, L. ervsimi is a harmful insect on Brassica oilseeds, especially on B. juncea in India and in other tropical regions of the world, causing up to 83% yield loss (Sekhon and Åhman, 1993; Mandal et al., 1994 as cited by Chattopadhyay et al., 2005; Agarwala and Datta, 1999; Aslam and Ahmad, 2001; Dutta et al., 2005; Hossain et al., 2006 and references therein). Aphids reproduce at a higher rate during the early vegetative stage of mustard plants when the developmental period is shortest and the production of winged morphs is lowest (Agarwala and Datta, 1999). The nymphs and adults cause damage by sucking away the plant sap, often covering the entire surface of the shoots, floral buds and pods. The pest breeds parthenogenetically, and an individual female gives birth to nymphs, which grow very fast and are completely bred in 7–10 days. About 45 generations are completed in a year. The high propagation rate of the pest affects the crop vitality, because the flowers fail to bear healthy pods, subsequently producing seeds of poor quality (Hossain et al., 2006 and references therein).

## 1.3 Plant Defence Mechanisms – General Information

Typical defence mechanisms that exist or are expressed in plants include constitutive, inducible, induced, direct and indirect defences. The existence, expression and functioning mechanism of these defences show both parallelism and contrasts to each other. These defence mechanisms are defined and addressed in this section, and are referred to and documented in other sections of the article with regard to the chemistry of Brassicaceae plants towards insect interactions.

Plants and animals both possess the potential to differentiate between self and non-self (hostile organisms), which may vary according to the heredity and environment. A major difference between plants and animals is that plants are sessile and animals mobile and the latter therefore can spread infections more easily (Jones and Takemoto, 2004). In order to overcome infections, plants are not only equipped with diverse constitutive/innate/preformed but also adapted defence mechanisms, to defend themselves. Likewise, striking similarities and obvious differences have evolved in animals which also form the basis of inducible or induced defence mechanisms (Nürnberger and Brunner, 2002; Montesano et al., 2003). Constitutive defence mechanisms, which are also regarded as ancient defence systems, are weapons that involve various receptors that recognise classes of microbial cellsurface molecules, the related signal transduction pathways that activate transcription of genes related to host defence, and the ubiquitous cationic peptides and proteins that act as antimicrobial effectors (Boman, 1995; Borregaard et al., 2000; Thomma et al., 2002).

An ecosystem comprises plants circumvented by herbivorous insects that continuously affect plant fitness. In order to overcome these herbivorous insects or to protect themselves from damage by herbivoral arthropods, plants have developed physical and chemical defences, which can either be innate or inducible in response to a certain attack (Takabayashi and Dicke, 1996; Karban and Baldwin, 1997; Paré and Tumlinson, 1999; Dicke and Hilker, 2003). The occurrence of both defence responses establishes an intricate network of defences for plants against insect herbivores. Inducible defences may provide an adaptive defensive strategy in which non-lethal cues from predators, herbivores or parasites provide a reliable indicator about the future risk of attack (Agrawal et al., 1999). In order to reduce the impact of insect attacks, plants have developed different defence strategies that include chemical and physical barriers such as induction of defensive proteins, volatiles that attract predators of the insect herbivores, toxic secondary metabolites, and

trichome morphology and density (Mello and Silva-Filho, 2002). These defence components produced by plants act both as constitutive substances to repel herbivores through direct toxication, or by lowering the digestibility of plant tissues, and as inducible substances produced in response to tissue damage by herbivores.

Induced defences are activated in the presence of an enemy, and then emerge or develop to their full strength. This particular kind of defence is important with regard to innate resistance, in case the defence is metabolically expensive, and when the attack is unpredictable but frequent (Haukioja, 1999). Plant responses that affect herbivore arthropods directly through systemic production of toxic metabolites are named direct defences, whereas responses that result in the attraction of natural enemies of the herbivores are designated as indirect defences (Dicke, 1999; Mattiacci et al., 2001). The term indirect defence, that is generally used when plants attract, nourish or house other organisms to reduce pressure from their enemies, has been referred to in the literature only in the last 20 years (Dicke and Sabelis, 1989; see review by Heil, 2008). Indirect defence mechanisms contribute towards the efficiency of the natural enemies of herbivores, e.g. through the emission of blends of volatile compounds and other secondary metabolites (Vet and Dicke, 1992; Hilker and Meiners, 2002; Kessler and Baldwin, 2002; Dicke et al., 2003; Rohloff and Bones, 2005). Such release of volatile compounds from plants has been considered as the cry or call for help by the plant from the carnivorous enemies (which are predators of herbivores) that might assist in reducing damage to plants. The release of volatile compounds from damaged plants derives from at least three biosynthetic pathways: first, the fatty acid (or octadecanoid) pathway that produces green leaf volatiles and (Z)-jasmone; second, the shikimic acid pathway that produces indole and methyl salicylate; and third, the isoprenoid pathway which produces terpenes (Hilker and Meiners, 2002). Indirect defences are mostly referred to as an environmentally-friendly crop protection strategy, but their plant fitness effects require more information in order to understand their ecological and evolutionary relevance before trophic interactions can be used as a reliable tool in agriculture (Heil, 2008).

## 2 Defence Components of Brassicaceae: Glucosinolates, the Glucosinolate-Myrosinase System, Plant Volatiles, Phytoalexins, Phytoanticipins and Sulphur

## 2.1 Glucosinolates

The characteristic feature of the Brassicaceae family is their production of specific secondary metabolites, the so-called glucosinolates (anionic thioglucosides) (Fahey et al., 2001; Bones and Rossiter, 2006). The glucosinolates constitute a large group of non-volatile and sulphur-containing secondary plant metabolites, which occur in all economically important Brassica crops (Tripathi and Mishra, 2007). Their known number totals almost 140 structures to date, 30 of which are present in Brassica species (Bellostas et al., 2007). Glucosinolates are  $\beta$ -thioglucoside N-hydroxysulphates with at least two sulphur atoms, one originating from cysteine, the other from phosphoadenosine phosphosulphate, possessing a  $\beta$ -D-glucopyranose moiety and a side chain derived from the amino acids methionine, tryptophan or phenylalanine (Fahey et al., 2001; Wittstock and Halkier, 2002; Rausch and Wachter, 2005). The content and composition of glucosinolates varies depending on Brassica species, the cultivar plant parts within the same plant, agronomic practices and climatic conditions (Sang et al., 1984; Clossais-Bernard and Larher, 1991; Rangkadilok et al., 2002; Font et al., 2005; Tripathi and Mishra, 2007). Glucosinolates are known to mediate interactions between Brassicaceae and their associated insect herbivores. They have been recognised as a class of natural pesticides since they exhibit toxic or repellent effects, by which they establish a significant defence mechanism to protect Brassica plants against pests and diseases (Mithen, 1992; Zukalová and Vašák, 2002). Glucosinolate levels and proportions of individual glucosinolate compounds have been demonstrated to be altered due to the damage caused by several insect pests (Koritsas et al., 1991; Bodnaryk, 1992; Hopkins et al., 1998). The glucosinolate concentration can increase in response to herbivore feeding, and this high level of glucosinolates can affect both generalist and specialist herbivores, and

glucosinolates can be equally effective as stimulants as well as deterrents (Bodnaryk, 1992; Bartlet et al., 1999; Li et al., 2000; Rask et al., 2000; Agrawal and Kurashige, 2003; Kuśnierczyk et al., 2007, 2008; Gols et al., 2008a).

## 2.2 The Glucosinolate-Myrosinase Defence System

The term myrosin cell, initially used by (Guignard, 1890), was discovered by (Heinricher, 1884). The myrosin cells can be easily distinguished from their neighbouring cells by light, electron and confocal microscopic observations (Bones and Iversen, 1985; Thangstad et al., 1990, 1991; Bones et al., 1991; Thangstad et al., 2004; Kissen et al., 2009). The myrosin cells contain less lipids, a high content of endoplasmic reticulum and harbour smooth-looking protein bodies referred to as myrosin grains (Bones and Iversen, 1985; Thangstad et al., 1991). Myrosin cells exist as scattered cells in stems, leaves, seeds, seedlings, petioles and roots. Brassica plants contain the enzyme myrosinase ( $\beta$ -thioglucoside glucohydrolase, thioglucosidase, EC 3.2.3.147 (formerly EC 3.2.3.1) (Bones and Slupphaug, 1989; Bones, 1990; Bones and Rossiter, 1996, 2006), which is thought to be exclusively present in myrosin cells (Thangstad et al., 1990, 1991; Bones et al., 1991; Höglund et al., 1991; Husebye et al., 2002; Thangstad et al., 2004; Kissen et al., 2009). In brassicas, myrosinases can be divided into three different gene families; the MA, MB and MC families (Xue et al., 1992; Chadchawan et al., 1993; Lenman et al., 1993; Thangstad et al., 1993; Falk et al., 1995). Furthermore, myrosinases are glycosylated dimeric proteins with subunit molecular weights from 62 to 75 kDa (Bones and Slupphaug, 1989; Bones and Rossiter, 1996). Myrosinases that belong to the MA family occur as free dimers (140 kDa), while members of the MB and MC families are found in high molecular complexes (200–1000 kDa), with myrosinase binding proteins (MBP) and the myrosinase associated proteins (MyAP) (Rask et al., 2000).

Glucosinolates and myrosinases are spatially segregated (Kelly et al., 1998; Koroleva et al., 2000; Husebye et al., 2002), but insect herbivory or tissue damage bring them together, which facilitates glucosinolate hydrolysis into thiocyanates, isothiocyanates, nitriles, oxazolidine-2-thiones and epithionitriles, depending upon pH and other conditions (Fig. 4) (Pivnick et al., 1992; Bones and Rossiter, 1996, 2006; Wittstock and Halkier, 2002).

The biosynthesis of glucosinolates still poses questions with, e.g., respect to transport and tissue, cell and subcellular spatial separation and organisation (Svanem et al., 1997; Thangstad et al., 2001; Halkier and Gershenzon, 2006). Glucosinolates such as progoitrin, sinigrin, gluconapin and glucobrassicanapin can give rise to cyanoepithioalkanes (epithionitriles). The exact mechanism behind the formation of epithionitriles from these glucosinolates is not known, but depends on the presence of a protein known as epithiospecifierprotein (ESP) (MacLeod and Rossiter, 1985; Foo et al., 2000; Zabala et al., 2005; Bones and Rossiter, 2006). The breakdown products resulting from glucosinolate hydrolysis represent the 'defence-active' components and the dual functioning of glucosinolates and myrosinases coming into contact upon tissue disruption is designated as the glucosinolate-myrosinase defence system. This system has been shown to have multiple roles in plant-insect interactions and insect pest management (Bones and Rossiter, 1996, 2006; Rask et al., 2000).

It should be noticed that some insects such as *B*. *brassicae* and *L*. *erysimi* actively take advantage of the defence compounds produced by their host plants by sequestering toxic compounds from the plant and using these compounds to protect themselves from predators. *B*. *brassicae* and *L*. *erysimi* sequester glucosinolates which can be degraded by a thioglucosidase endogenously produced by the insect, when the latter is crushed (Jones et al., 2001, 2002; Bridges et al., 2002; Rossiter et al., 2003; Husebye et al., 2005). These crushed insects likely both smell/taste badly and release volatiles, alarming other aphids in the colony.

## 2.3 Plant Volatiles

Throughout evolution, higher plants have evolved complex mechanisms to be able to communicate with their environment. Based on their capability for gas exchange, plants can release mixtures of plant volatiles having distinct biological functions related to plantinsect, plant-pathogen and plant-plant communication, and adaption to stresses (Kishimoto et al., 2005;



Fig. 4 Insect herbivory brings glucosinolates and myrosinase together and facilitates the hydrolysis of glucosinolates

Baldwin et al., 2002). Brassicaceae produce volatile and semi-volatile toxic compounds based on glucosinolate degradation upon tissue damage, thus directly functioning in plant defence. However, plant volatile communication is a much more sophisticated process (described in Sect. 3.2) of trophic interactions where, e.g., herbivore-attacked plants release volatile signals to attract predators of the feeding insects. In order to

Compound group	Plant volatiles	Plant organ	Function
Green leaf volatiles	C <sub>6</sub> -alcohols, aldehydes and acetates	Green plant parts	Plant-plant signalling, predator attraction, antimicrobial activity
Plant hormones	Jasmonic acid and salicylic acid derivatives, ethylene	Whole plant	Plant-plant signalling, induction of plant defences
Terpenes	Mono- and sesquiterpenes	Flowers, leaves, roots	Flower pollinator attraction, attraction of predators, antimicrobial activity
Aromatics	Benzyl and phenylethyl- derivatives	Mainly flowers	Flower pollinator attraction, antimicrobial activity
Glucosinolate-derived volatiles	Isothiocyanates, thiocyanates, oxazolidine thiones, nitriles, epithionitriles	All plant parts containing myrosinase and glucosinolates	Plant defence, herbivore attraction
Sulphur-containing compounds	Sulphides, elemental sulphur	Probably whole plant	Plant defence

Table 3 The diversity of plant volatiles and signalling compounds in Brassicaceae plants (Rohloff and Bones, 2005)

distinguish between the multiple roles and functions of plant volatiles in Brassicaceae, Table 3 briefly summarises the known chemical structure groups found in different plant parts and tissues.

#### 2.4 Phytoalexins and Phytoanticipins

Phytoalexins are low-molecular-weight antimicrobial compounds or secondary metabolites that are synthesised de novo, while phytoanticipins are pre-formed inhibitors of infection (Dixon, 2001; Rouxel et al., 1991). However, the distinction between phytoalexin and phytoanticipin is not always clear as some compounds may be phytoalexins in one species, and phytoanticipins in others (Dixon, 2001). Phytoalexins and phytoanticipins are also referred to as two significant classes of natural pesticides exerting different methods of action (Zukalová and Vašák, 2002). Phytoanticipins emerge from already created precursors, and phytoalexins commence as the result of an external affect due to distinct metabolic activity. Glucosinolates and the glucosinolate-myrosinase system represent an example of such a type of phytoanticipin since myrosinase and glucosinolates are already biosynthesised as precursors before insect attack (Zukalová and Vašák, 2002). The glucosinolate-myrosinase system, which has been throughly investigated in the past decades, is to a large extent considered as a constitutive, as well as inducible type of defence system. It is highly dynamic, interactable with insect pests and a wellestablished mechanism towards integrated pest management (Bones and Rossiter, 1996, 2006; Rask et al., 2000; Wittstock et al., 2004; Müller and Sieling, 2006). Isothiocyanates produced after glucosinolate hydrolysis by myrosinases play crucial ecological roles in protecting plants against various pests, including insects and microbial systems. Therefore, isothiocyanates are part of a group of basic plant chemical defences known as phytoanticipins (Pedras et al., 2007a). Moreover, phytoalexins from the Brassicaceae family are the only sulphur-containing and nitrogen-containing phytoalexins including an unexpected range of functional groups and indolyl structures (Pedras et al., 2007b). Brassinin, 1-methoxy brassinin, brassilexin and cyclobrassinin are sulphur-containing indole phytoalexins, which have been isolated from different Brassica species (Rouxel et al., 1991). Brassinin and 1-methoxybrassinin, which contain a dithiocarbamate group, were the first phytoalexins to be reported. Dithiocarbamates have been recognised as important pesticides and herbicides and until now crucifers are the only plants known to produce such compounds (Pedras et al., 2000).

### 2.5 Sulphur

The plants of the family Brassicaceae are known to be rich in sulphur (Williams and Cooper, 2004). Sulphur is necessary for plant development and sulphurcontaining compounds such as sulphur-rich antifungal proteins, phytoalexins and glucosinolates play an important role in plant defence against pathogens (Dubuis, 2004). Brassica plants use sulphur (S) to synthesise glucosinolates and phytoalexins. Cysteine, the primary product of sulphur assimilation, is incorporated into sulphur-rich proteins (SRPs; including thionins) and glutathione. Furthermore, cysteine is the donor of reduced sulphur for glucosinolate biosynthesis and for the synthesis of phytoalexins (including camalexin) (Rausch and Wachter, 2005). Low sulphate availability has also been shown to induce the expression of myrosinase proteins in Sinapis alba plants (Bones et al., 1994; Visvalingam et al., 1998). Pathogen attack and abiotic elicitors lead to the synthesis of sulphur-containing phytoalexins such as brassinin and concentration at the site of pathogen attack. Moreover, leaves of some Brassica varieties possess the constitutive elemental sulphur  $(S^0)$  that may be related to an alternative process of S<sup>0</sup> biosynthesis, such as from the degradation of certain glucosinolates (see reviews Bones and Rossiter, 1996; Williams and Cooper, 2004). Elemental sulphur as cyclooctasulphur S<sup>8</sup> has been reported to have antimicrobial activity in Theobroma cacao (Cooper et al., 1996) and similar inorganic sulphur compounds are also present in Brassicaceae (Rohloff and Bones, 2005).

## 3 Chemical Ecology and Insect Behavioural Aspects

The chemical ecology of plant-insect interactions deals with chemical signals mediating all aspects of insects' lives, their ecological interactions through identification, and defining the chemicals involved in these interactions (Cardé and Millar, 2004). These chemical signals and ecological interactions include: chemical mediators modifying insect behaviour, plant chemicals to protect from insect herbivores (below- and above-ground), multitrophic interactions among plants-herbivores-parasitoids-hyperparasitoids, oviposition, semiochemical-mediated interactions, and the chemical cues that parasitoids use to find their herbivore hosts. Plant chemicals that elicit immediate behavioural responses in insects are generally categorised as attractants (eliciting oriented movements towards the source), arrestants (causing aggregation), stimulants (eliciting feeding, phagostimulation, oviposition, etc.), repellents (causing oriented movements away from the source) and deterrents (inhibiting feeding or oviposition) (Ryan, 2002).

In an article entitled "Developing sustainable pest control from chemical ecology" (Pickett et al., 1997), the significance of chemical ecological relations is emphasised in order to understand insect/insect and insect/plant interactions, and insect behaviour influenced by pheromones and other semiochemicals. This type of knowledge opens up promising pest control methods as alternative strategies to the exclusive use of broadspectrum pesticides. In general, insect behaviour results from the integration by its central nervous system of a variety of inputs that derive from stimuli acting on exteroceptors (that sense events external to the insect), enteroceptors (that sense the internal physiological state of the insect), and proprioceptors (that sense the relative positions of parts of the body) (Foster and Harris, 1997).

On the background of chemical ecological and insect behavioural aspects, we have mainly focused on and reviewed regulation of oviposition during insect attack and behaviour of the different herbivore insects that attack brassicas, and their trophic interactions. The oviposition and trophic interaction studies mainly revolve around the response of plant cues and chemicals from *Brassica* plants towards these insects, which have been discussed with special emphasis on glucosinolates, their breakdown products, and blends of plant volatiles that are released during oviposition and multitrophic (second, third or fourth level) interactions. Figures 5 and 6 show the diagrammatic presentation of larvae feeding, oviposition process by adult insects, the trophic interactions of Brassica plants as hosts, their herbivore insects and the predators or parasitoids of herbivores.

## 3.1 Oviposition

The search for an oviposition site by flying insects comprises two phases of behaviour, i.e. pre-alighting and post-alighting (Hopkins et al., 1999). The prealighting behaviour of an insect, captivated by a range of visual and odour stimuli, ends up in contact with a potential host plant, and the post-alighting behaviour of the female on the host plant depends upon the balance of the internal physiological condition and



**Fig. 5** *Brassica* plant as host showing leaf feeding, oviposition and trophic interactions by larvae and adults of *Pieris brassicae* and *Delia radicum* and their parasitoids (modified after Ross-Craig, 1949)

external stimuli that she perceives (Hopkins et al., 1999). Olfaction (the sense of smell) has an important role after an insect has landed on a plant, and before it moves to leaves or down to the soil to oviposit. Physical factors such as size, colour and leaf structure characteristics have been shown to influence oviposition behaviour, since bright green surrogate leaves (sprayed

with a leaf surface extract of the host plant, with a stem, possessing vertical folds to mimic veins, and with a paraffin cover to mimic the wax layer) received the most eggs (Roessingh and Städler, 1990). Most herbivorous insects firstly attack plants by oviposition. Most lepidopterans do not feed on leaves as adults, but females deposit their eggs on those plants or plant parts



Fig. 6 Brassica plant as host showing aphid oviposition, colony and trophic interactions between cabbage aphid (Brevicoryne brassicae) and its ladybird predators (Coccinella septempunctata) (modified after Ross-Craig, 1949)

where hatching larvae will find suitable food (Hilker and Meiners, 2006). In order to protect themselves against feeding damage, the oviposition-induced plant responses are targeted against eggs laid on the plant, the hatching larvae, or the egg-laying gravid female (Hilker and Meiners, 2002, Hilker and Meiners, 2006).

Chemical stimulation of oviposition is a complex process, and specific chemicals are involved in the acceptance of hosts and rejection of non-host plants (Hamilton et al., 2005). Ovipositing females seem to employ plant volatiles as cues for orientation to host plants and the following contact evaluation of plants by means of less- or non-volatile secondary metabolites has a great significance in host recognition (Keiichi, 1995). It was further highlighted that the acceptance or rejection of a particular plant by females is regulated not only by the presence or absence of oviposition stimulants, but by negative stimuli evoked by co-occurring deterrents. Moreover, oviposition-deterring activity has also been observed

**Table 4** Definitions of significant terms mentioned in the article

Term		Definition	Reference
1	Allelochemical	An infochemical that mediates an interaction between two individuals that belong to different species.	(Dicke and Sabelis, 1992)
2	Antixenosis	A term that is derived from the Greek word 'xeno' (guest) that describes the inability of a plant to serve as host to an arthropod, e.g. insect, represents plant traits conferring non-preference of herbivores, i.e. reduced acceptance for oviposition or feeding.	(Smith, 2005)
3	Deterrent	A chemical that inhibits behaviour, such as feeding or oviposition, when applied to a site where such behaviour normally occurs.	(Blossey and Hunt-Joshi, 2003)
4	Infochemical	A chemical that, in the natural context, conveys information in an interaction between two individuals, evoking in the receiver a behavioural or physiological response.	(Dicke and Sabelis, 1992)
5	Semiochemical	A term that is derived from the Greek word 'semion' (a mark or a signal). Chemicals, which function in communication between and among species, as well as those that serve as messengers between members of the same species.	(Law and Regnier, 1971 as cited by Paré and Tumlinson, 1999)

in leaves without eggs, but adjacent to those carrying eggs, hence indicating a systemic effect (Hilker and Meiners, 2002). Plant secondary compounds, influencing antixenosis (Table 4), are mentioned as a link with oviposition specificity or feeding stimulation of more than 20 insect species (Hopkins et al., 1997). Moreover, plant secondary compounds (or allelochemicals) (Table 4) from a particular plant may be oviposition stimulants or deterrents for insects, which feed on that plant (Renwick and Radke, 1981, 1985) and oviposition deterrence by these compounds possibly has significant consequences for crop pest management (Tabashnik, 1987). These chemical deterrents/stimulants evidently play a significant role in the acceptance or rejection of plants as hosts by ovipositing female butterflies.

It has been stated that the induced defence responses both in above-ground and below-ground plant parts are common (Van Dam and Raaijmakers, 2006). Feeding damage by below-ground herbivores may cause a systemic increase in defensive compounds in above-ground parts, hence leading to interactions between above-ground and below-ground herbivores feeding on the same plant. These kinds of aboveand below-ground interactions stimulated by induced responses may alter damage patterns and, finally, affect fitness or plant survival (Bezemer and Van Dam, 2005; Van Dam and Raaijmakers, 2006). Furthermore, the root herbivores of oilseed rape, kale, swede and canola have been shown to affect plant performance, commercial yield and plant chemical defence. Based on oviposition behaviour of different insects and plant defence, we have reviewed essential works that have been conducted on insects that oviposit on *Brassica* plants. In recent years, most of the investigations pertaining to insect oviposition behaviour and related chemical aspects of *Brassica* plants have been focused on white butterflies (*Pieris* spp.), cabbage seed and seedpod weevils (*Ceutorhynchus* spp.), and cabbage root and turnip root flies (*Delia* spp.), with few studies on diamondback moth (*P*. xylostella), cabbage looper (*T*. ni), cabbage moth (*M*. brassicae) and brassica pod midge (*D*. brassicae).

# 3.1.1 White Butterflies (*Pieris brassicae* and *P. rapae*)

Almost thirty years ago, it was observed that gravid female *P*. *brassicae* butterflies show discrimination behaviour in their choice of a plant to oviposit on (Behan and Schoonhoven, 1978). In another study from the 1980s regarding oviposition preferences of field-collected *P. rapae* butterflies towards *B. nigra* plants, leaf water content was considered as a phenotypic characteristic associated with oviposition preference (Wolfson, 1980). Renwick and Radke (1985) further showed that oviposition by *P. rapae* on cabbage was deterred by homogenised cabbage tissue sprayed onto intact plants. Polar as well as non-polar extracts of non-host plants inhibited oviposition. In another investigation, oviposition by

P. rapae butterflies was deterred by spraying the plant secondary compounds coumarin and rutin on cabbage plants in greenhouse tests. Both coumarin and rutin deterred oviposition primarily by affecting prealighting rather than postalighting behaviour, indicating that deterrence was mediated by non-contact cues (Tabashnik, 1987). In a comparative study of oviposition responses of P. rapae and P. napi to nine crucifers, the results showed that the two Pieris spp. have apparently evolved differential sensitivities to the chemical stimuli that trigger or deter oviposition. The balance of positively and negatively interpreted sensory signals generated by plant chemicals obviously plays an important role in acceptance or rejection of a plant by both species. Moreover, in an investigation of leaf volatiles, while the large white butterfly (P. brassicae) was laying eggs, it was found that the young cabbage leaves on which P. brassicae had laid eggs emitted larger amounts of the monoterpene  $\alpha$ -thujene than young clean leaves (Bergström et al., 1994). Furthermore, the oviposition-deterring pheromones (ODPs) of P. brassicae and P. rapae have been considered to be produced in the female accessory glands, and contain volatile and non-volatile components. Oviposition-deterring pheromones are natural and species-specific compounds that inhibit oviposition, and seem suitable chemicals to reduce crop infestation by insects utilising them, since they produce very low environmental risks (Schoonhoven, 1990).

The potential of glucosinolates in stimulating feeding by larvae and oviposition by adults has been confirmed for *P*. *brassicae* and *P*. *rapae* through studies showing that ovipositing *P*. *rapae* adults respond more strongly to indole glucosinolates, such as glucobrassicin, and less strongly to aliphatic glucosinolates, such as glucocheirolin (Rodman and Chew, 1980; Renwick et al., 1992; Van Loon et al., 1992; Huang and Renwick, 1993; Renwick, 2001). Some of these studies showing response of these insects towards glucosinolates and isothiocyanates are described in more detail below.

Already in 1980, Rodman and Chew showed that the oviposition and larval feeding by P. *napi* is associated with glucosinolate profiles of plant species. Later, Klijnstra and Roessingh (1986) suggested that foretarsal taste hairs of females, apart from the glucosinolate cells, also possess sense cells specifically sensitive to the oviposition-deterring pheromones. These different sense cells are responsible for the sensitivity of tarsal "B-type hairs" to eggwash and glucosinolates. The so-called "B-type hairs" comprise one type of trichoid sensilla which is recognised on the tarsi of P. brassiace (Ma and Schoonhoven, 1973). The oviposition by *P*. rapae on cabbage was shown to be stimulated by the glucosinolate glucobrassicin. Other studied glucosinolates were sinigrin, which was slightly active, and glucoiberin, which was completely inactive as a stimulant (Renwick et al., 1992). Another investigation involving oviposition bioassays between cabbage (B. oleracea) leaves and P. brassicae also identified glucobrassicin as the oviposition stimulant (Van Loon et al., 1992). Moreover, Traynier and Truscott Traynier and Truscott (1991) showed that solutions of glucobrassicin purified from foliage and sinigrin elicited oviposition by the cabbage butterfly, P. rapae, at threshold concentrations as low as  $10^{-6}$  M while at higher concentrations, glucobrassicin elicited a faster oviposition rate and a stronger visual response to the substrate through associative learning. In another comparative study on the relative activities of 10 glucosinolates in stimulating oviposition by P. rapae and P. napi, it was observed that in most cases, P. rapae was more sensitive to aromatic and indole glucosinolates than to aliphatic ones (Huang and Renwick, 1994); however, P. napi responded strongly to aliphatic glucosinolates.

(Bruinsma et al., 2007) investigated the effect of jasmonic acid (a key hormone involved in plant defence responses) on Brussels sprouts (*B. oleracea* var. *gemmifera*) and their acceptability for oviposition by the butterflies *P. rapae* and *P. brassicae*. The investigation showed that both butterfly species laid fewer eggs on leaves of jasmonic acid-treated plants than on leaves of control plants. It was further shown that application of jasmonic acid doubled the concentration of the glucosinolate glucobrassicin and lowered the concentrations of the glucosinolates glucoiberin and 4-hydroxyglucobrassicin.

### 3.1.2 Cabbage Seed and Cabbage Seedpod Weevils (*Ceutorhynchus assimilis* and C. *obstrictus*)

In order to develop resistant crop germplasm, it is important to understand how host-plant characteristics affect behavioural and physiological responses of insect herbivores. With this hypothesis, Ulmer and Dosdall (2006) investigated feeding, oviposition preference, larval development and oviposition behaviour of the cabbage seedpod weevil (*C. obstrictus*) on eight Brassicaceae species that differ in their glucosinolate profiles. Among these eight *Brassica* species, the preferred host plant for feeding and oviposition was *B. carinata*, larval development occurred most rapidly on *B. rapa*, and the larval weight was highest on *B. napus*. Total glucosinolate levels did not influence *C. obstrictus* larval growth or development; however, high levels of specific glucosinolates such as sinalbin and gluconapin were associated with increased developmental time or reduced weight.

In a linear track olfactometry test to observe responses of the cabbage seed weevil (C. assimilis) to volatiles from oilseed rape (B. napus), weevils showed attraction towards the odour of rape during a short period before diapause and for most of their postdiapause life (Bartlet et al., 1993). Apart from that, attraction was also observed for 3-butenyl and 4-pentenyl isothiocyanate, but not for 2-phenylethyl isothiocyanate. A mixture of the three isothiocyanates was more attractive than the individual isothiocyanates. Furthermore, the effects of extracted and artificial oilseed rape odours on the behavioural response of male and female cabbage seed weevils were investigated in a wind tunnel experiment (Evans and Allen-Williams, 1998). Omission of two isothiocyanates from the artificial extract significantly reduced the upwind movement of females.

After oviposition into a pod of oilseed rape, the female cabbage seed weevil marks the pod with oviposition-deterring pheromones by brushing it with her eighth abdominal tergite (Ferguson et al., 1999b). This oviposition-deterring secretion of the cabbage seed weevil was demonstrated to contain isoand n-alkanes, dimethylalkanes, alkenes, fatty acids, 15-nonacosanone, 15-nonacosanol, and cholesterol (Mudd et al., 1997). Extracts of volatiles entrained from ovipositing weevils failed to inhibit oviposition. The authors evidenced that the oviposition-deterring pheromones of the cabbage seed weevil are sensed primarily by contact chemoreception at the sensilla chaetica of the antennae, and the electrophysiological responses recorded from these gustatory sensilla retain significance as a bioassay to assist identification of the active constituent(s) of the pheromone (Ferguson et al., 1999a). In their previous study, it was reported that the decision of a seed weevil to accept or reject a

pod for oviposition to some extent depends upon cues perceived via antennal sensilla. In order to select an oviposition site, a female weevil walks back and forth along the pod, antennating the substrate much more intensively than when walking on the stem or petiole (Ferguson and Williams, 1991).

# 3.1.3 Cabbage and Turnip Root Flies (Delia radicum and D. floralis)

The very first detailed work published on the behavioural responses of cabbage root fly (D. radicum) (Traynier, 1967a, b) to host plants specified that the odour stimulates activity and that chemical contact stimulates oviposition (Hawkes and Coaker, 1979). In a visual host finding, and shape recognition study on the cabbage root fly, using four shapes of yellow sticky traps as plant models, Tuttle et al. (1988) indicated that yellow discs or crosses at ground level baited with allyl isothiocyanate effectively monitored female cabbage root fly. The main factors affecting female landing were suggested to be the colour of substrate, height above the ground, presence of host volatiles in the vicinity, visual prominence, and size of the total area of 'attractive' colour. The initial landing phase of turnip root fly (D. floralis) (period between landing on the leaf and the first movement across the leaf) was shown to be crucial for host-plant expedition prior to oviposition site selection (Hopkins et al., 1997). The behavioural sequence analysis of individual gravid female turnip root flies showed that during the postalighting behaviour of turnip root fly, the decision to reject a highly resistant plant was predominantly based on plant cues received during a stationary period immediately after landing on the leaf (the leaf contact phase) (Hopkins et al., 1999). The host-plant acceptance by the cabbage root fly seems to result from a synergistic response to simultaneously perceived olfactory and contact chemostimulation (De Jong and Städler, 1999; De Jong et al., 2000). It was elucidated that female flies, after landing on a potential host, explore the plant by walking on the leaf surface and stem. Due to this exploration, the taste receptors located on the tarsi detect the presence of host-specific cues which stimulate the flies to descend to the soil and oviposit.

Glucosinolates, glucosinolate hydrolysis products such as isothiocyanates, and other volatile compounds are important cues for the cabbage root fly in locating and recognising a suitable host, and are involved in the feeding behaviour. Only mated gravid females respond to them (Hawkes and Coaker, 1979; Nottingham and Coaker, 1985; Renwick et al., 1992; Roessingh et al., 1992; Simmonds et al., 1994; De Jong and Städler, 1999; Hurter et al., 1999). In a study examining the role of glucosinolates towards oviposition behaviour of the cabbage root fly using egg counts and electrophysiological recordings from tarsal chemoreceptors, Roessingh et al. (1992) showed that the D sensilla on segments 3 and 4 of the tarsus of cabbage root fly females contain a sensitive receptor cell for glucosinolates. The flies clearly distinguished between model leaves with and without glucosinolates, but a clear dose response curve was only obtained for the indole glucosinolate glucobrassicin, which indicates that glucobrassicin on the cabbage leaf surface stimulates oviposition. Another study involving behavioural and chemosensory responses of the turnip root fly to glucosinolates demonstrated that slight modifications in chemical composition of glucosinolates resulted in alterations in neural activity (Simmonds et al., 1994). Furthermore, of the eleven glucosinolates tested in these studies, the flies reacted most to glucobrassicanapin, gluconapin and glucobrassicin. Investigation of B. nigra and B. oleracea plants against cabbage root fly displayed a local increase in indole glucosinolates in the main roots, with B. oleracea plants showing a stronger increase in indole glucosinolate levels than B. nigra, which was the preferred feeding site of cabbage root fly larvae (Van Dam and Raaijmakers, 2006). Moreover, the increase in indole glucosinolates in B. nigra main roots was counterbalanced by a significant decline in aromatic glucosinolates.

Apart from glucosinolates, *Brassica* phytoalexins have been documented as playing a significant role in oviposition (Roessingh et al., 1997; Baur et al., 1996, Baur et al., 1998; De Jong et al., 2000; Hurter et al., 1999; Marazzi et al., 2004a, b). Two compounds, the so-called "Cabbage Identification Factors" (CIFs) are isolated from the surface of *B. oleracea* cv. *botrytis* leaves and identified. These compounds, perceived by a specific receptor neuron in the tarsal sensillum C5 of the female fly (Roessingh et al., 1997), are strong stimulators of oviposition in the cabbage root fly. Spectroscopic data indicated that the main CIF compound (1,2-dihydro-3-thia-4,10,10btriaza-cyclopenta[.a.]fluorene-1-carboxylic acid) is a novel compound related to Brassica phytoalexins such as brassicanal C and is accompanied by its glycine conjugate. Cabbage (B. oleracea) leaves contain these compounds in extremely low concentrations but higher levels were detected in the roots of B. napus var. napobrassica (rutabaga) (De Jong et al., 2000). Furthermore, surrogate leaves treated with methanolic leaf surface extracts of *B*. napus plants that received three different sulphur fertilisation treatments showed even more marked differences by the oviposition choice of cabbage root fly than the potted plants. The oviposition data was shown to be positively correlated either with CIF or glucosinolates (Marazzi et al., 2004a). Investigation on application of eleven crucifer-specific phytoalexins and related synthetic compounds on surrogate paper leaves, being offered to cabbage root flies in oviposition assays, showed three of them (methoxybrassinin, cyclobrassinin and brassicin) to be significantly stimulatory, whereas the remaining metabolites had no effect, suggesting that the reaction of the fly appears to be structure-specific (Baur et al., 1998). In their previous observations, the authors suggested that the cabbage root fly tends to choose plants infected by suitable bacteria, and thus phytoalexins produced by the infected plant might be providing the specific guiding signal (Baur et al., 1998). The cabbage root fly seems to present another example of a herbivore preferring already attacked plants, which is consistent with the possibility that some phytoalexins act either as a signal for host-plant detection or as a marker for optimally preconditioned plants.

### 3.1.4 Diamondback Moth (Plutella xylostella)

The oviposition preference of *P. xylostella* for cabbage, broccoli and cauliflower was observed in the field (Hamilton et al., 2005). No difference was observed in the number of eggs found on the broccoli or cauliflower cultivars. The eggs that were laid on cultivar *Savoy King* were higher than any of the cultivar tested. Larvae development was more rapid and longer on the cabbage cultivar *Green Coronet* than the cultivar *Savoy King*. It was concluded that in the field, *Savoy King* (cabbage cultivar) is more attractive to oviposition.

Renwick et al. (2006) demonstrated isothiocyanates to be oviposition stimulants for the diamondback moth. Hughes et al. (1997) showed that potent oviposition stimulants for the diamondback moth are extracted from cabbage foliage by soaking the intact leaves in chloroform. Analysis of these extracts revealed the presence of two isothiocyanates, iberin and sulphoraphane. Other isothiocyanates with sulphur in the side chain were also reported to be active.

In 1960, Gupta and Thorsteinson studied the effect of sulphur mineral nutrition on two Brassicaceae species (S. alba and B. nigra) and demonstrated that the constituents of host plants affected larval feeding and oviposition of P. xylostella. In order to observe sulphur perception by the diamondback moth, Marazzi et al. (2004a) raised B. napus plants under three different sulphur regimes: sulphur-free, normal field concentration and sulphur-rich (twice the normal field concentration). In addition, they performed dual oviposition assays with the diamondback moth, using either Brassica plants or artificial leaves sprayed with methanolic leaf-surface extracts. It was shown that chemical compounds on the leaf surface mediate the oviposition preference and that the female insect can perceive the quality of the host plants in terms of their fertilisation status. Since the leaf content of the volatile isothiocyanates is influenced by sulphur nutrition, the authors analysed the extracts for the presence of these compounds. Eleven glucosinolates were identified, with progoitrin and gluconapoleiferin being the most abundant ones. Sulphur nutrition has also been reported to differentially affect the expression of the glucosinolate-hydrolysing myrosinases (Bones et al., 1994).

#### 3.1.5 The Cabbage Looper (Trichoplusia ni)

The cabbage looper exhibits a chemical spacing mechanism. This chemical spacing mechanism was noticed as feeding larvae deterred oviposition by gravid females and larval frass, since debris or excrement produced by insects were found to contain the biologically active material (Renwick and Radke, 1980 as cited by Renwick and Radke, 1981). The concept that host-plant chemicals play a role in the spacing of phytophagous insects was introduced by Cirio in 1971 (as cited by Renwick and Radke, 1981). Based on this concept, Renwick and Radke (1981) also demonstrated that the cabbage looper depends on the host plants to avoid overcrowding. Furthermore, the results by Landolt (1993) suggested an important role for damage-induced plant volatiles in host location as well as host acceptance by *T*. *ni*.

Chow et al. (2005) observed the effects of larval experience with complex plant latex (Hoodia gor*donii*) on subsequent feeding and oviposition by the cabbage looper moth. The study showed that naïve groups of T. ni moths are deterred from feeding and ovipositing on cabbage leaves treated with the H. gordonii latex, and that larval feeding experience can lessen or reverse this deterrence. It was further suggested that moths may be acquiring oviposition preferences from larval feeding experience as described by Hopkins' host selection principle (HHSP). This principle, also called theory of larval memory, postulates that the adult females of phytophagous insects will prefer to feed or oviposit on the same plant species upon which they themselves developed as larvae (Hopkins, 1917 as cited by Barron, 2001 and stated by Chow et al., 2005).

### 3.1.6 The Cabbage Moth (*Mamestra brassicae*) and Bertha Armyworm (*Mamestra configrata*)

The cabbage moth (M. brassicae) often chooses Brassica plants as hosts for oviposition (Ulland et al., 2008). The mortality of eggs, larvae, pupae and larval dispersal of the cabbage moth on white cabbage (B. oleracea var. capitata) was investigated in a series of small-scale field experiments and in the laboratory (Johansen, 1997). The highest mortality was found in young larvae and in hibernating pupae. The main mortality factor was found to be unfavourable weather conditions, and cold stress.

Furthermore, in other oviposition studies, Rojas and Wyatt (1999) analysed the influence of pre-imaginal (larval conditioning) and post-imaginal experience (adult conditioning) on the orientation, landing and oviposition of the female cabbage moth in a wind tunnel. The females were initially attracted to and landed on chrysanthemum whether or not they fed on this plant species. In addition, the oviposition preference for cabbage plants was not changed by the larval feeding regimen (a regulated system of a diet). Overnight exposure of females to chrysanthemum or cabbage plants decreased the subsequent orientation/landing on the same species offered in non-choice tests. However, it had no effect on subsequent oviposition as females

from both treatments oviposited more on cabbage. In another study, Rojas et al. (2000) investigated the orientation and oviposition behaviour of M. brassicae on the most preferred host (cabbage) and two other host plants, tomato and chrysanthemum. It was observed that after landing on the plant, the insects were most likely to lay eggs on cabbage and tomato, but the behavioural sequence on these plants was shown to be different. Half of the females laid eggs on tomato only after dragging the ovipositor on the leaf, whereas almost all females laid eggs on cabbage after touching the surface with the ovipositor only briefly. The authors proposed that an understanding of these behaviours could help establish appropriate conditions for future studies on chemical identification of plant semiochemicals mediating host finding of the cabbage moth. The same group also investigated the age at which females begin to lay, the daily pattern of oviposition and the influence of host-plant material on egg-laying, as well as the host preference of M. brassicae under laboratory conditions (Rojas et al., 2001). It was observed that the females started ovipositing during the third, fourth and fifth scotophases after emergence. Maximum oviposition occurred during the second hour of scotophase. The females without host-plant material laid fewer eggs than females with host-plant material (B. oleracea var. capitata). In two-choice tests, the females preferred to oviposit on cabbage rather than chrysanthemum, but there was no difference in the mean number of eggs laid on cabbage and tomato.

Ulland et al. (2008) identified methyl salicylate as primary odorant of a specific receptor neuron type, and showed that it inhibits oviposition by M. brassicae. The behavioural effect of methyl salicylate was studied in outdoor test arenas with B. napus and artificial plants (Ulland et al., 2008). The experiments indicated that mated M. brassiace females avoid plants with dispensers emitting methyl salicylate. It was further pointed out that as methyl salicylate is induced by caterpillar feeding, it may mediate a message to mated M. brassicae females that the plant is already occupied.

The oviposition biology of the Bertha armyworm was studied in relation to the effect of conspecific eggs on oviposition site selection (Ulmer et al., 2003). Females strongly preferred to oviposit on leaves with eggs of a different female than on leaves without eggs. Gravid females preferred leaves that were treated with methanol, highlighting that the source of oviposition stimulation is possibly chemical-based.

## 3.1.7 The Brassica Pod Midge (Dasineura brassicae)

The oviposition behaviour of the brassica pod midge was compared on a preferred host (B. napus) with that of a non-preferred, less suitable host (B. juncea) for larval growth (Åhman, 1985). The number of landing females was significantly higher on B. napus than on B. juncea, indicating host differences in olfactory and/or visual stimuli. After landing, the females showed different behaviour on the two species by staying longer and laying more egg batches on B. napus than on *B. juncea* plants (Åhman, 1985). In another study by the same author looking at the toxicities of Brassica secondary compounds to the eggs of D. brassicae, the nitrile compound 1-cyano-2-phenylethane was shown to be more toxic than a "green leaf alcohol", (Z)-3-hexen-1-ol (Åhman, 1986). The author further suggested that a crucifer specialist may be restricted in its use of particular hosts due to the compositions and concentrations of glucosinolate compounds.

## 3.2 Trophic Interactions Among Brassica Crops-Herbivores-Parasitoids-Hyperparasitoids

"Nearly 75% of the world's macroscopic biodiversity is tied up in the look between plants, herbivores, predators and decomposers. In this context, the study of trophic interactions, involving plants, herbivores, and their predators or parasitoids represents a frontier in ecology, and this knowledge can be integrated in environmentally sound agricultural pest management" (Sergio Rasman, Cornell University, USA).

Multitrophic interactions among host plants (first trophic level), herbivores (second trophic level), natural enemies of herbivores/carnivorous arthropods/parasitoids/predators (third trophic level), and hyperparasitoids (fourth trophic level) have been well documented (Agrawal, 2000; Shiojiri et al., 2002; Ode, 2006; Heil, 2008). Parasitoids and predators of herbivores have evolved and generally perform their activities within a multitrophic framework. Plants emit diverse blends of volatile compounds from their leaves, flowers and fruits, which affect a range of organisms in the environment including pollinators, herbivores, neighbouring plants and carnivores (Takabayashi and Dicke, 1996; Neveu et al., 2002; Shiojiri et al., 2002). The emission of these volatiles has mutualistic effects being clearly beneficial for the plant such as in the attraction of pollinators to flowers. Several investigations and documentations from tritrophic level interactions among plants, herbivores, and parasitoids or predators have illustrated that these components are tightly interwoven. Interactions between plants and natural enemies can be antagonistic, additive or synergistic, which illustrates the importance of multitrophic perspectives for effective and sustainable pest management strategies (Wright and Verkerk, 1995; Gange and Brown, 1997; Lewis et al., 1997; Tscharntke and Hawkins, 2002; Gripenberg and Roslin, 2007; De Boer et al., 2008). Apart from bi- or tritrophic, the infochemicals released by an infested plant and/or herbivores are available to other trophic levels. The same information used by the parasitoid to locate the herbivore may be utilised by hyperparasitoids (the fourth trophic level) in order to locate the parasitoids.

Plants under herbivore attack produce chemical cues due to the mechanical damage. These chemical cues are important signals for orientation of both carnivorous enemies and herbivores, including distant host location by arthropods (Karban and Baldwin, 1997; Dicke, 1999; Dicke and Van Loon, 2000; Arab and Bento, 2006). Plants' responses upon damage caused by herbivores occurs through the regulation of several biochemical pathways that lead to the release of chemical compounds, which cause either repellence to herbivore insects or attraction to natural enemies (predators or parasitoids) of herbivore insects (Karban and Baldwin, 1997; Dicke and Van Loon, 2000; Pickett et al., 2003; Arab and Bento, 2006; Gols and Harvey, 2009). Attracted by herbivory-induced plant volatiles, parasitoids of herbivores perform host searching to lay their eggs in or on them as shown in recent studies (Mattiacci et al., 1994; Takabayashi and Dicke, 1996; Dicke, 1999; Hilker and Meiners, 2002; Fatouros et al., 2005b; Heil, 2008). Apart from being chemical cues for parasites and predators, these socalled semiochemical volatiles may induce defence responses in neighbouring plants (Paré and Tumlinson,

1999). Semiochemicals emitted from a diverse group of plants and insects mediate key processes in the behaviour of specific insects.

In addition, several ecological approaches have emphasised the significance of plant traits in plant-insect interactions (Agrawal, 2000), which may vary due to genetic variations among plants and/or induced responses in individual plants upon herbivore attack. These effects are determined mainly by nutritional quality, physical structure, defence-related volatiles and other secondary metabolites. Furthermore, nonlethal exposure of an animal to carnivores, and a plant to herbivores, not only induces a defence, but causes the attacked organisms to produce offspring that are better defended than offspring from untreated parents. This is generally referred to as the transgenerational effect (Agrawal et al., 1999) and likely involves epigenetic modifications. The transgenerational induction of defences has been described as a new level of phenotypic plasticity across generations which might be an important component of predator-prey interactions (Agrawal et al., 1999).

Moreover, the tritrophic role of plant chemistry is a key to various aspects of trophic phenomena (Ode, 2006). This includes top-down effects (controlled by predators) versus bottom-up effects (controlled by resources) and enemy-free space and host choice. The tritrophic effects of plant chemistry are valuable to assess the degree of compatibility between biological control and plant resistance approaches to manage pests (Ode, 2006). Therefore, the study of trophic interactions and their manipulation has the potential to lead to effective ways of biological pest control, and thus reduce pesticide use (Agrawal, 2000; Dicke et al., 1990). Some of the multitrophic (second, third or up to fourth level) studies on Brassica species will be reviewed below. Most of these studies revolve around one single herbivore species such as Pieris spp., P. xylostella, Delia spp. and aphids, but studies investigating different herbivore insects simultaneously are also described.

#### 3.2.1 Parasitoids or Predators

*Cotesia rubecula* Marshall is a solitary endoparasitoid of the small white butterfly (*P. rapae*) and *Cotesia* glomerata L., a gregarious parasitoid of both *P. rapae* and *P. brassicae*. *P. rapae* and *P. brassicae* are also the hosts for the egg parasitoid Trichogramma evanescens Westwood in cabbage (Noldus and Van Lenteren, 1985a, b). Cotesia plutellae Kurdjumov is a dominant solitary koinobiont, larval endoparasitoid of *P. xylostella*, and generally regarded as being highly specific to P. xylostella (Talekar and Shelton, 1993; Agelopoulos and Keller, 1994a-c; Geervliet et al., 1994, 1998; Mattiacci et al., 1994, Mattiacci et al., 2001; Harvey et al., 2003; Fatouros et al., 2005b). Diadegma semiclausum Hellén is a specialised parasitoid of P. xylostella (Bruinsma et al., 2009). Platygaster subuliformis Kieffer and Omphale clypealis Thompson are among the more than 20 species of hymenopteran parasitoids that attack D. brassicae (Murchie et al., 1997). Lysibia nana Gravenhorst is a solitary hyperparasitoid of newly cocooned prepupae and pupae of several microgastrine braconids, including C. glomerata (Harvey et al., 2003). Adalia bipunctata L. (predator) is known for its polyphagy against many aphid species (Hodek, 1959, as referred to by Francis et al., 2001). The seven-spot ladybird Coccinella septempunctata L. is recorded as the only parasitoid of *B. brassicae* (Acheampong and Stark, 2004). Diaeretiella rapae M'Intosh, a predominant parasitoid of Brassica-feeding aphids, attacks the mustard aphid L. erysimi at a greater rate than the generalist feeding aphid *M*. persicae (Blande et al., 2007). The parasitoid Trybliographa rapae Westwood is a specialist larval endoparasitoid of D. radicum (Neveu et al., 2002). Phradis interstitialis Thomson, Phradis morionellus Holmgr. and Tersilochus heterocerus Thomson are among the most frequent pollen beetle parasitoids (Jönsson et al., 2005). Among these three parasitoids, the parasitoid P. morionellus attacks larvae inside oilseed rape buds and flowers and also feeds on the flowers (Jönsson and Anderson, 2007 and references therein). Microplitis mediator Haliday is an important parasitoid of early instar larvae of the cabbage moth M. brassicae (Lauro et al., 2005). The root-lesion nematode (Pratylenchus penetrans Cobb) is a migratory endoparasite with a broad host range (Baldridge et al., 1998 and references therein).

### 3.2.2 Host Plants (*Brassica* spp.)-Herbivores (*Pieris* spp.)-Parasitoids-Hyperparasitoids

Among the four different crucifer-specific compounds tested towards perception of *P. rapae*, phenylace-

tonitrile elicited a higher response than ally1 isothiocyanate, benzyl isothiocyanate and 2-phenylethyl isothiocyanate (Hern et al., 1996). Furthermore, *P. rapae* is considered to be an important pollinator for many plant species, since the butterfly occurs three to four times a year and tends to visit a diversity of flowers (Ômura et al., 1999). As an innate preference, butterflies are attracted by specific aromatic volatiles from rape flowers such as benzaldehyde, phenylacetaldehyde, benzyl alcohol, 2-phenylethanol, phenylacetonitrile and indole, in decreasing order of quantity (Ômura et al., 1999). Moreover, experience-based food consumption studies with larvae of P. rapae have shown that plant host preference is dependent on the abundance of glucosinolates in brassicas, independent of the chemical structure of these compounds (aliphatic or aromatic) (Renwick and Lopez, 1999). The role of glucosinolates acting as feeding stimulants for larvae of the large white butterfly P. brassicae was recognised long ago (Verschaffelt, 1910) as mentioned by David and Gardiner (1966). David and Gardiner (1966) in their *P. brassicae* and glucosinolate-related study demonstrated that P. brassicae reared on fresh cabbage leaf to the end of the fourth instar would not accept diets containing sinigrin (and sucrose) in the fifth instar. Another interesting point is that although glucosinolates act as feeding stimulants for P. brassicae larvae, about 50% of unfed, newly hatched larvae will accept a diet which contains no glucosinolates. It was noted that out of the nine effective glucosinolates, four (glucoiberin, glucoerucin, sinigrin and progoitrin) were from cabbage. Agrawal and Kurashige (2003) analysed the classical interaction between P. rapae and isothiocyanates. Using whole plants, root extracts and a microencapsulated formulation of allyl isothiocyanate, it was shown that isothiocyanates reduce herbivore survival and growth, and increase development time, each in a dose-dependent manner. Neither the substrate allyl glucosinolate, nor myrosinase negatively affected *P. rapae*, hence presenting strong evidence for a role for isothiocyanates in plant resistance against the specialist herbivore *P*. rapae.

Karowe and Schoonhoven (1992) determined the relative suitability of *Brassica* as host plants both for unparasitised *P*. *brassicae* caterpillars and for *C*. *glomerata* developing in *P*. *brassicae*. Of all these *Brassica* plants (Brussels sprouts, Swedish turnip and rape varieties of *B*. *napus*), the host-parasitoid complex attained a lower final weight than unparasitised
*P. brassicae*, probably due to reduced consumption by the parasitised *P. brassicae*. In contrast, Sato and Ohsaki (2004) elucidated that *C. glomerata*, although a potential parasitoid of *P. brassicae*, did not effectively lower the population density of *P. brassicae* immediately after the first invasion due to its reluctance to parasitise.

The introduction of a new species can alter the attributes of other species within a community, which may affect discontiguous trophic levels via adjacent trophic levels (Tanaka et al., 2007). The exotic large white butterfly P. brassicae invaded Hokkaido Island, Japan, and quickly spread throughout the island. Prior to the invasion, the small white butterfly P. rapae was the host of the primary parasitoid C. glomerata, on which both the larval hyperparasitoid Baryscapus galactopus and the pupal hyperparasitoid Trichomalopsis apanteroctena depended. At the time of the invasion, C. glomerata generally laid eggs exclusively in *P. rapae*. During the five years following the invasion, however, the clutch size (number of eggs laid in a single nesting) of C. glomerata in P. rapae gradually decreased, whereas the clutch size in P. brassicae increased. As a consequence, the invasion of P. brassicae changed the host use of the primary parasitoid C. glomerata and the pupal hyperparasitoid T. apanteroctena within a very short time (Tanaka et al., 2007).

In a study assessing the attractive role of infochemicals originating from either the host, P. brassicae, or its food plant, cabbage, it was shown that C. glomerata responds to chemical signals emitted from herbivoredamaged plants rather than infochemicals from P. brassicae (Steinberg et al., 1993). Geervliet et al. (1997) demonstrated that the parasitoids of Pieris species, C. glomerata and C. rubecula, showed differential responses towards various herbivoreinfested food plants. Herbivore-infested plants emit a huge number of compounds to their maximum level, and the parasitoids that search for hosts have to deal with variability in the availability of chemical cues emitted by the food plants and their host. In a tritrophic study with P. brassicae and C. glomerata, Mattiacci et al. (1994) investigated the significance of herbivory-induced plant volatiles in Brussels sprouts leaves (B. oleracea) versus infochemicals released by the larvae. Chemical analysis of the headspace of undamaged, artificially damaged, caterpillar-infested and caterpillar regurgitant-treated leaves showed that

the plant responds to damage with an increased release of volatiles. Another tritrophic study revealed that the solitary parasitoid C. rubecula discriminated among volatile blends from Brussels sprouts plants treated with regurgitants of unparasitised P. rapae or P. brassicae caterpillars over blends emitted by plants treated with regurgitant of parasitised caterpillars (Fatouros et al., 2005b). The parasitoid C. glomerata discriminated between volatiles induced by regurgitant from parasitised and unparasitised caterpillars of its major host species, P. brassicae. Another study of a system comprising Brassica-Pieris-Trichogramma suggested that egg deposition induces alteration in plant surface chemicals, thus causing the arrest of egg parasitoids by contact cues around the eggs (Fatouros et al., 2005a).

Studies by Blaakmeer (1994) regarding infochemicals in a tritrophic system of Brassica, Pieris and Cotesia identified the glucosinolate glucobrassicin as an oviposition stimulant. In another tritrophic study, Blaakmeer et al. (1994) analysed headspace composition, collected either from intact cabbage plants or cabbage plants infested with either P. brassicae or P. rapae first instar larvae. They identified twenty-one volatiles in the headspace of intact plants. The major differences between intact and caterpillar-damaged plants regarding the headspace profile were revealed for hexyl acetate, (Z)-3-hexenyl acetate, myrcene, sabinene and 1,8-cineole. The larval endoparasitoid C. glomerata was attracted by the volatiles released from B. oleracea damaged by P. brassicae first instar larvae. C. rubecula, a specialised larval endoparasitoid of P. rapae, was attracted by the volatiles released from the B. oleracea-P. rapae plant-host complex. This shows that cabbage plants kept under the conditions of headspace collection produce attractive volatiles for both parasitoids. Furthermore, Harvey et al. (2003) examined the interactions over four trophic levels. The study involved B. oleracea and a naturally occurring population of *B*. *nigra* because of the difference in their glucosinolate content. The concentrations of glucosinolates were more than 3.5 times higher in young shoots of B. nigra than in corresponding shoots of B. oleracea. The study observed that the cocoon mass of C. glomerata was not affected by the host-plant species on which P. brassicae fed; however, L. nana survival was greater and the body size was larger when P. brassicae fed on B. oleracea. It was demonstrated that the qualitative differences in herbivore diet can differentially affect the performance of interacting organisms across several trophic levels with a proposition that the bottom-up forces may also play a role in mediating interactions involving plants-herbivores-parasitoids and hyperparasitoids (Harvey et al., 2003).

In a perspective paper, Dicke et al. (2004) presented a tritrophic system comprising host cabbage plants, herbivorous larvae of P. brassicae and the parasitoid C. glomerata. The damage caused by caterpillars feeding on cabbage plants differentially regulates the expression of various genes in the plants, up-regulates biosynthesis of certain types of glucosinolates, and emits bouquets of volatile organic compounds (VOCs). These VOCs are shown to act as an indirect defence by attracting parasitoids that laid eggs in the caterpillars. Among these volatile organic compounds, the greenleaf volatile (Z)-3-hexen-1-ol and the terpenoid 1,8cineole were shown to be the main volatiles emitted by the cabbage plants. Mumm et al. (2008) investigated the significance of terpenoids in a 'cabbage' system consisting of Brussels sprouts plants, large and small cabbage white butterflies (P. brassicae and P. rapae), and the larval parasitoid C. glomerata. The terpenoid emission was manipulated by treating the plants with fosmidomycin, which inhibits one of the terpenoid biosynthetic pathways, and consequently terpenoid emission. The study demonstrated that inhibitors such as fosmidomycin can be used to investigate the role of terpenoid infochemicals in plant defence mechanisms against herbivores.

## 3.2.3 Host Plants (*Brassica* spp.)-Herbivore (*P. xylostella*)-Parasitoids

Olfactory attraction of female diamondback moths (P.xylostella) to the odours of intact and homogenised host plants was investigated using behavioural and electrophysiological methods (Pivnick et al., 1994). Allyl isothiocyanate from *B. juncea* and *B. napus* plants was the most attractive component, being absent in odours from intact plants. It was further suggested that certain elements of this fraction, possibly in combination, are important olfactory cues for host-plant finding by the diamondback moth, with isothiocyanates playing an important and synergistic role, particularly when plants are damaged. Van Loon

et al. (2002) showed that the diamondback moth larvae employ a combination of biosynthetically distinct categories of feeding stimulants which allows for a higher degree of discriminatory ability than glucosinolates alone.

Karimzadeh and Wright (2008) used a tritrophic crucifer-*P*. *xylostella-C*. *plutellae* experimental system in order to test the hypothesis of host-plant effects challenging the innate immune system of an insect host. Using measures of the two principal immune effectors against parasitoids, encapsulation and phenoloxidase activity, it was shown that despite having strong plant effects on parasitism, parasitoidal effects on immune effectors of the host were transitory. These varied levels of parasitism of *P*. *xylostella* mediated by plant quality are stated to be an outcome of behavioural and fitness factors rather than a reduced immune challenge.

With a consideration that the parasitoids Trichogramma chilonis and C. plutellae, and the predator Chrysoperla carnea are the potential biocontrol agents, Reddy et al. (2002) conducted olfactory response studies with P. xylostella on cabbage plants. Among the four larval frass that were tested, only allyl isothiocyanate elicited significant responses in the parasitoids and predator, but C. plutellae and C. carnea responded well to all four volatiles (dipropyl disulphide, dimethyl disulphide, allyl isothiocyanate and dimethyl trisulphide). The results indicated that the sex pheromone and larval frass volatiles from the diamondback moth and volatile compounds from cabbage may be used as natural enemies to locate diamondback moth (Reddy et al., 2002). Furthermore, the same group performed another study on the host plant-mediated orientation and oviposition by the diamondback moth and its predator Chrysoperla carnea in response to four different Brassica host plants: cabbage, cauliflower, kohlrabi and broccoli (Reddy et al., 2004). The results indicated that the orientation of female diamondback moths and C. carnea females towards cabbage and cauliflower was significantly greater than toward broccoli or kohlrabi plants. Furthermore, in free-choice tests, oviposition by the diamondback moth was significantly greater on cabbage, followed by cauliflower, broccoli and kohlrabi, while C. carnea preferred to oviposit on cabbage and cauliflower, followed by broccoli and kohlrabi.

## 3.2.4 Host Plants (*Brassica* spp.)-Herbivores (*M. brassicae* and *P. brassicae*)-Parasitoids

Lauro et al. (2005) examined the attack responses of female M. mediator to the first three larval instars of M. brassicae. Their results suggested that first and second instar larvae of M. brassicae are suitable hosts for M. mediator. Third instar larvae are suboptimal because of unsuccessful oviposition attempts and immature parasitoids failed to complete development. However, naïve attacking parasitoids exhibited minimal discrimination among instars, although experienced parasitoids most frequently attacked first instar larvae.

In a four-armed airflow olfactometry study with the egg parasiotoid T. evanescens, Noldus and Van Lenteren (1985a) observed that the females were attracted by a volatile substance (or substances) released by virgin females of cabbage white butterfly (P. brassicae). T. evanescens was also observed to be attracted by the volatiles released by calling virgin cabbage moths, M. brassicae. The results of the experiments with M. brassicae showed that T. evanescens is attracted by volatiles which are probably the sex pheromone released by calling virgin females, but not by the main component (Z)-11-hexadecenylacetate, nor by a crude extract of the sex pheromone gland (at the concentrations tested). The study suggested the significance of volatile kairomones for the egg parasite T. evanescens in the host-habitat location. In another follow-up study, Noldus and Van Lenteren (1985b) showed that contact kairomones are involved in host location by T. evanescens. Indeed, T. evanescens females searched significantly longer on cabbage leaves treated with the wing scales of two hosts, P. brassicae and P. rapae. Furthermore, egg washes of P. brassicae containing an oviposition deterrent pheromone for the butterflies were found to have a contact-kairomonal effect on the parasite.

Gardner et al. (2007) performed an oviposition experience study to observe the response of egg parasitoid (T. vanescens) towards contact kairomones of two different host species M. brassicae and P. brassicae. The response of T. evanescens was influenced by the number of eggs it had laid, but oviposition did not result in a significant change in behaviour. Parasitoids readily accepted an egg of a second species, and the time spent searching in a particular kairomone area appeared to depend on the reproductive state and expected survival of a parasitoid rather than the development of any host preference. It was further proposed that T. *vanescens* seems to select patches on the basis of reward probability rather than maximising reward size.

## 3.2.5 Host Plants (*Brassica* spp.)-Herbivore (*A. rosae*)-Parasitoids

Müller and Arand (2007) tested whether adults of A. rosae innately prefer the plant species for oviposition that allows the best larval performance with regard to both developmental conditions and (plant-derived) defence efficiency against predators. It was observed that in A. rosae innate preferences of ovipositing, females seem to be mainly influenced by the host-plant effects on larval developmental times rather than potential defence efficiency of larvae against predators. The study demonstrated that for preference and performance studies, not only the plant-insect interactions should be considered, but also the biotic determinants (Müller and Arand, 2007). In a previous study by the same group, Müller and Brakefield (2003) tested the role of the A. rosae sawflies' haemolymph and of the glucosinolate sinalbin in the defence against wasps by using manipulation assays. The haemolymph released by easy bleeding and its chemical components proved to be in part responsible for the efficient defence of the sawfly against wasps (Müller and Brakefield, 2003).

## 3.2.6 Host Plants (*Brassica* spp.)-Herbivores (*Delia* spp.)-Parasitoids

Birch et al. (1992) showed that different *Brassica* genotypes such as kale, rape and swede after inoculation with turnip root fly (*D. floralis*) eggs showed a similar response in glucosinolate metabolism. Total glucosinolate content in roots increased due to a two-to fourfold increase in indole-based compounds. The largest increase for an individual glucosinolate after attack was found for glucobrassicin. Root damage did not significantly affect stem glucosinolate composition but resulted in an increase in aliphatic glucosinolates with a corresponding decrease in indole-based compounds in leaves. Furthermore, Griffiths et al. (1994) investigated induced changes in the indole glucosinolate content of oilseed and forage rape (*B. napus*)

plants in response to either turnip root fly (*D. floralis*) larval feeding or artificial root damage. Larval damage increased the total glucosinolate content of the roots. In contrast, artificial damage reduced the total glucosinolate content of the roots. It was also interesting to note that the concentration of glucobrassicin present in the oilseed rape cultivar *Ariana* was consistently higher in both the larval and artificially root-damaged plants compared with undamaged plants of the same cultivar.

The long-range plant involvement of volatiles in host location of the herbivore D. radicum on the parasitoid T. rapae was investigated by Neveu et al. (2002). It was shown that the infested turnips systemically emit herbivore-induced plant volatiles that attract the parasitoid T. rapae, implying that the production of parasitoid-attracting volatiles appeared to be systemic in this particular tritrophic system.

## 3.2.7 Host Plants (*Brassica* spp.)-Herbivore (*D. brassicae*)-Parasitoids

The responses of *D. brassicae* and its parasitoids *P. subuliformis* and *O. clypealis* to allyl- and 2-phenylethyl isothiocyanates were investigated using traps in winter oilseed rape (Murchie et al., 1997). In this study it was observed that the traps baited with allyl isothiocyanate caught more male and female *D. brassicae* and more female *O. clypealis* than traps baited with 2-phenylethyl isothiocyanate caught more male and female *P. subuliformis* than traps baited with 2-phenylethyl isothiocyanate caught more male and female *P. subuliformis* than traps baited with 2-phenylethyl isothiocyanate caught more male and female *P. subuliformis* than traps baited with allyl isothiocyanate caught more male and female *P. subuliformis* than traps baited with allyl isothiocyanate or unbaited traps.

## 3.2.8 Plants (*Brassica* spp.)-Herbivores (*Phyllotreta* spp. and *Meligethes aeneus*)-Parasitoids

Already in 1956, Görnitz showed that flea beetles are attracted to allyl isothiocyanate, which is a volatile hydrolysis product of the glucosinolate sinigrin. Based on these results, Pivnick et al. (1992) performed trapping experiments to test attraction of crucifer-feeding flea beetles to volatile glucosinolate hydrolysis products released from glass vials. The pattern of attraction was the same for the flea beetle species *P. cruciferae* and *P. striolata*. When captures of the two species were pooled, 3-methylthiopropyl isothiocyanate, methyl isothiocyanate and n-butylisothiocyanate were found to be significantly attractive, while nitriles were the least attractive compounds. Flower and leaf extracts of rape were found to be attractive in the field from at least 20 m, and the pollen beetles were proposed to use odour-mediated upwind anemotaxis to locate oilseed rape plants (Evans and Allen-Williams, 1994). The results from Y-tube olfactometer bioassays indicated that *M*. *aeneus* is able to locate its host plant by olfactory stimuli in the early bud stage, i.e. the stage at which the infestation begins in the field and when the typical yellow colour and floral scent of oilseed rape are absent (Ruther and Thiemann, 1997). In a field study on the effect of trap design, trap colour and isothiocyanate lures on the capture of the pollen beetle, M. aeneus was shown to be attracted to four alkanyl, three alkenyl and 2-phenylethyl isothiocyanate lures (Blight and Smart, 1999). The response of M. aeneus to yellow water traps baited with individual lures of 25 floral volatile compounds was also investigated in 17 field experiments with B. napus (Smart and Blight, 2000). These compounds comprised seven nitrogenous amino acid derivatives, five non-nitrogenous amino acid derivatives, nine fatty acid derivatives and four isoprenoids. The results showed most compounds to be attractive, but four fatty acid derivatives were repellent. 1-Hexanol was either attractive or repellent, depending on the release rate. It has been further suggested that because of its polyphagous nature, M. aeneus responds to a large number of chemically diverse compounds. The attraction to the most effective unbaited yellow traps was enhanced 1.7-3.3 times with the addition of a lure comprising a mixture of allyl, 3butenyl, 4-pentenyl and 2-phenylethyl isothiocyanate.

Jönsson et al. (2005) analysed behavioural responses in three ichneumonid pollen beetle (M. aeneus) parasitoids (P. interstitialis, P. morionellus and T. heterocerus) to volatiles emitted from different phenological stages of oilseed rape. All three parasitoid species were attracted to odours from the bud stage of oilseed rape. T. heterocerus was attracted to odours of flowering rape, but the two Phradis species avoided the flower odours. However, when the odours of flowering rape were in combination with yellow-coloured flowers, and odours of the bud stage were in combination with green-coloured buds, P. interstitialis was equally attracted to both stimuli, and T. heterocerus showed an increased preference for flower odours, while no effect of colours could be found in P. morionellus. The terpenes sabinene, myrcene, limonene and (E, E)- $\alpha$ -farnesene were the dominant volatiles in the bud and flower headspace. A group of aromatic compounds including benzaldehyde, methyl benzoate and phenyl acetaldehyde were mainly released from the flowering rape. Furthermore, Jönsson and Anderson (2007) investigated emission of oilseed rape volatiles after pollen beetle infestation by analysing behavioural and electrophysiological responses in the parasitoid P. morionellus. It was observed that both starved and fed parasitoids preferred infested rape, but the proportion of responding female P. morionellus was significantly lower for the group that was starved. Six of the 20 volatiles identified were released at higher rates from infested than from non-infested rape. The volatiles released at a significantly higher rate from infested rape and detected by P. morionellus antennae were (Z)-3-hexenylacetate, (Z)-3-hexenol, 3-butenyl isothiocyanate and (E, E)- $\alpha$ -farnesene.

## 3.2.9 Plants (Brassica spp.)-Herbivores (Ceutorhynchus assimilis)

In a study analysing the peripheral olfactory perception of isothiocyanates, cabbage seed weevil (C. assimilis) antennal receptors were able to locate a broad range of compounds associated with the odour of its host plant (Evans and Allen-Williams, 1992). Green leaf volatiles and specific host-related compounds were demonstrated to play a significant role in overall perception and recognition of host odour. In a response study with cabbage seed weevil antenna to volatiles in air entrainment-derived extracts of oilseed rape, using electroantennograms coupled with GC-single cell recording, it was shown that isothiocyanate perception is mediated by three types of olfactory cells, which show differential response to 3-butenyl, 4-pentenyl and 2-phenylethyl isothiocyanate (Blight et al., 1995). The responses of cabbage seed weevil to other electrophysiologically-active volatiles from rape were tested in a linear track olfactometer (Bartlet et al., 1997). Attraction was shown to nitriles (phenylacetonitrile, 4-pentenenitrile and 5-hexenenitrile) and to volatiles emitted by a wider spectrum of plant families, (Z)-3-hexen-1-ol and methyl salicylate. The combination of an isothiocyanate mixture with phenylacetonitrile increased attraction, but there was no such increase when the isothiocyanate mixture was combined with methyl salicylate. Furthermore, the response of the cabbage seed weevil to yellow water traps baited with some components of oilseed rape odour was studied in a series of field experiments (Smart and Blight, 1997). Four isothiocyanates, five other amino acid derivatives (aromatic compounds) and two fatty acid derivatives were detected by peripheral olfactory receptors of C. assimilis. The results suggested that yellow traps baited with 2-phenylethyl isothiocyanate can be used to monitor immigration into crops in the spring and that phenylacetonitrile may be useful for assessing numbers of weevils colonising the crop throughout the summer. After several field experiment studies on trap design, trap colour, and a mixture of isothiocyanates on the capture of C. assimilis, Smart et al. (1997) showed that the sticky card trap, mounted at 45° to the vertical and baited with the isothiocyanate mixture, may be useful for monitoring movement of C. assimilis during migratory periods.

## 3.2.10 Host Plants (*Brassica* spp.)-Aphids-Parasitoids/Predators

Chemicals that influence aphid behaviour have been studied for many decades, but a major breakthrough occurred in the early 1970s, due to chemical identification of an alarm pheromone for a number of species by several groups (as reviewed by Dawson et al., 1990). This pheromone was shown to comprise the sesquiterpene hydrocarbon (E, E)- $\alpha$ -farnesene. The chemical potential of glucosinolates and the glucosinolate-myrosinase system has been shown for cabbage and mustard aphids. The cabbage aphid, designated as the walking mustard oil bomb (Bridges et al., 2002; Jones et al., 2001, 2002; Kazana et al., 2007), is not only capable of sequestering harmful glucosinolates but also catalyses the hydrolysis of accumulated glucosinolates upon predator feeding in order to generate biologically active and toxic isothiocyanates. Both B. brassicae and L. erysimi produce an endogenous insect myrosinase, thus mimicking the plant glucosinolate-myrosinase system and its spatial organisation (Rossiter et al., 2003; Jones et al., 2001, 2002; Bridges et al., 2002; Husebye et al., 2002; Kazana et al., 2007). Recent studies with the model plant A. thaliana have shown the induction of aliphatic

glucosinolates (Mewis et al., 2006), the contribution of indole glucosinolates towards aphid deterrence of M. persicae (Kim and Jander, 2007), and alterations in metabolism and chemical defence by B. brassicae (Kuśnierczyk et al., 2007, 2008). It is likely that similar defence responses exist in important Brassica crops.

For the mustard aphid (L. erysimi), isothiocyanates together with (E, E)- $\alpha$ -farnesene have been reported to work as alarm signals (Dawson et al., 1987). The sequestration of glucosinolates by the specialist aphid, B. brassicae, may provide protective compounds against attack by predators or parasitoids. It was suggested that the functioning of glucosinolate utilisation may be important in understanding the exploitation of biological control agents such as predators and parasitoids to control specialist B. brassicae and generalist *M*. persicae aphid species (Cole, 1997). Electrophysiological recordings together with highresolution gas chromatography identified metabolites of glucosinolate precursors in the cruciferous host plants as being synergists for the alarm pheromone of L. erysimi. The most active, allyl isothiocyanate, significantly improved the activity of an aqueous formulation of  $(E, E) - \alpha$ -farnesene (Dawson et al., 1990). The volatile (Z)-jasmone was shown to repel aphids while being an attractant to parasitoids (Birkett et al., 2000). Feeding behaviour of the specialist B. brassicae and the generalist M. persicae on the range of Brassica species tested indicated that generalist and specialist aphids are influenced differently by the host plant (Cole, 1997). M. persicae did not generally accept or reject Brassica species due to the presence of phagostimulants, such as glucosinolates at the leaf surface or along the stylet pathway, unless the concentration was very high.

Francis et al. (2001) performed a tritrophic interaction study on different *Brassica* crops with variable glucosinolate profiles, with *M. persicae* and *B. brassicae* and the predator *A. bipunctata*. It was suggested that the pest management must include both herbivore and plant trophic levels to determine the plant allelochemical impact on the third trophic level, the beneficial entomophagous insects. It was further added that the semiochemicals from plants, directly or through herbivore prey, must be considered as a potential toxin or reliable infochemical in relation to the efficacy of pest control by natural enemies. In another study either *B. nigra* plants (characterised by high levels of sinigrin), or an artificial aphid diet to which sinigrin was selectively added, were used to rear the specialist B. brassicae and aphids were provided as a food source to two species of polyphagous ladybirds, A. bipunctata and C. septempunctata (Pratt et al., 2008). The results indicated that the presence of sinigrin in the diet of B. brassicae makes this aphid unsuitable as a food source for A. bipunctata, but not for C. septempunctata. However, there appear to be costs associated with C. septempunctata feeding on aphids that contain this secondary metabolite. Blande et al. (2007) investigated the orientation behaviour of the parasitoid D. rapae to the semiochemicals produced when the two aphid species L. erysimi (specialist) and *M. persicae* (generalist) were fed on turnip. Isothiocyanates were among the compounds emitted by Brassicaceae plants in response to insect feeding damage, including damage caused by aphids. The results suggested that similar cues may be utilised by L. erysimi and M. persicae for host location, whereas the acceptance of hosts and their suitability may involve aspects of non-volatile aphid chemistry. In an olfactory perception and orientation behaviour study of the aphid B. brassicae and the parasitoid D. rapae to alkenyl glucosinolate hydrolysis products, electroantennogram responses indicated peripheral odour perception in D. rapae females to all 3-butenylglucosinolate hydrolysis products (Pope et al., 2008). Nevertheless, rearing D. rapae either on B. nigra, which accumulates 2-propenylglucosinolate, or B. rapa, which accumulates 3-butenylglucosinolate, changed the parasitoids' response to 3-isothiocyanatoprop-1-ene and 4-isothiocyanatobut-1-ene.

## 3.2.11 Host Plants (*Brassica* spp.)-Herbivores [(*P. xylostella/Pieris*) (*P. xylostella/M. persicae*)]-Parasitoids and [*P. rapae* and Two Different Root Feeders]

Some of the tritrophic studies involving two herbivores documented below are based on the hypothesis suggesting that plants in both natural and cultivated ecosystems are subjected to insect infestation by more than one species and that induced responses occur both locally and systemically throughout the plant (Agelopoulos and Keller, 1994a–c; Shiojiri et al., 2000; Agbogba and Powell, 2007).

The role of volatiles in attracting C. rubecula to cabbage infested by the host P. rapae was elucidated (Agelopoulos and Keller, 1994b). The study showed that C. rubecula was attracted to cabbage previously infested by P. rapae. Additionally, females were also attracted to mechanically damaged cabbage and cabbage previously infested by P. xylostella (a non-host lepidopteran herbivore). Furthermore, another tritrophic study by the same group focused on the identification of the volatile compounds that could be involved in the searching behaviour of the parasitoid C. rubecula (Agelopoulos and Keller, 1994c). The volatiles emitted by the intact cabbage were  $\alpha$ -pinene,  $\beta$ -pinene, myrcene, 1,8cineole, n-hexyl acetate, (Z)-3-hexen-1-yl acetate and dimethyl trisulphide. Mechanical damage on an intact plant induced the release of two more compounds, (E) - 2-hexenal and l-methoxy-3-methylene-2-pentanone. Feeding by larvae of P. rapae induced the plant to release all the compounds upon mechanical damage and additionally 4-methyl-3-pentenal and allyl isothiocyanate. On the other hand, feeding by larvae of P. xylostella induced the plant to release all the compounds present after mechanical damage with addition of allyl isothiocyanate. Shiojiri et al. (2000) performed a comparative study of host-searching behaviour with two parasitoid species (C. plutellae and C. glomerata) on a cabbage plant, infested by the host larvae P. xylostella. It was found that the parasitoids showed their antennal-searching behaviour only on the host-infested site. The searching time of C. plutellae on a piece of cabbage leaf infested by host larvae was significantly longer than a piece infested by nonhost (P. rapae) larvae. Likewise, the searching time of C. glomerata on the host (P. rapae)-infested piece of cabbage was significantly longer than that on a leaf infested by non-host (P. xylostella) larvae. Furthrmore, Shiojiri et al. (2002) conducted another study by using the same system as previously (Shiojiri et al., 2000) confirming that the fitness effects of induced chemical production by plants should be studied in tritrophic interaction webs, because chemicals induced by one herbivore species may indirectly affect oviposition preferences of heterospecific herbivores by altering the effectiveness of their respective parasitoids. Such indirect interactions between host and non-host herbivores are referred to as infochemical-mediated indirect interactions. In a recent study, Bruinsma et al. (2009) showed that feeding by *P. rapae* and *P. xylostella* 

resulted in increased endogenous levels of jasmonic acid in B. oleracea (Brussels sprouts) plants. However, the levels of the intermediate 12-oxophyto-dienoic acid (OPDA) were induced only after P. rapae feeding. Moreover, jasmonic acid-induced volatiles of B. oleracea attracted parasitoids (C. glomerata, C. rubecula and D. semiclausum) in a time- and dose-dependent application (Bruinsma et al., 2009). In a study using two insects, the peach aphid (M. persicae) and the caterpillar (P. xylostella) on cabbage plants, Agbogba and Powell (2007) looked at the responses of the aphid parasitoid D. rapae by Y-tube olfactometry. Their results indicated that the aphid and the caterpillar induce different changes in the volatile profile of cabbage plants and that D. rapae females readily distinguish between the two. In a study, involving B. nigra plants, P. rapae and two different root feeders (the endoparasitic nematode P. penetrans and the larvae of the cabbage root fly D. radicum), Van Dam et al. (2005) indicated that root feeding can significantly alter the nutritional quality of shoots by changes in secondary metabolite levels and hence the performance of a specialist shoot feeder. Moreover, it was observed that *P*. rapae larvae grew more slowly and produced fewer pupae on plants that were infested with root feeders, especially on plants infested with the endoparasitic nematode P. penetrans and that both glucosinolate as well as phenolic levels were affected by root feeding.

# 4 Brassicas and Their Potential for Integrated Pest Management

Integrated pest management (IPM) is an ecological approach to managing insect pests, by using different pest control methods, that are aimed at the entire pest complex of a crop ecosystem and finally ensures high-quality agricultural production in a sustainable, environmentally safe, and economically sound manner (Bajwa and Kogan, 2002 as cited by Guerena, 2006). The insect pest management strategies include pest monitoring, that allows accurate timing of pesticide applications; combined use of semiochemicals, hostplant resistance and trap crops to manipulate pest behaviour, employing biological control approaches or selective insecticides to reduce pest populations and developing insect-resistant crops (Pickett et al., 1997). From these perspectives of pest management aiming at an ecochemical control, we have briefly reviewed the use of brassicas as biocontrol, cover crops and trap crops in order to control or manipulate the insect-pest complex not only for brassicas, but also insect pests and diseases for other economically important crops. In addition, we have described in brief the role of plant breeding in developing insect resistance in *Brassica* crops.

## 4.1 Brassicas as Biocontrol Agents

Brassicas as biocontrol agents generally employ the process of biofumigation. Biofumigation is referred to as the process of suppression of pests and pathogens through release of volatile substances from degradation of biomass into the soil. For the biofumigation process, Brassica plants are incorporated into the soil and due to the significant amounts of glucosinolates, toxic breakdown products, especially isothiocyanates, are produced after tissue maceration (Brown and Morra, 1997; Kirkegaard and Sarwar, 1998; Gardiner et al., 1999; Smith et al., 2004). Isothiocyanates carry fumigant properties similar to metham-sodium (Sarwar et al., 1998; Warton et al., 2003). Metham-sodium (sodium N-methyldithiocarbamate) is a compound that is used globally to control noxious soil-borne organisms in intensive cropping systems as it generates the wide-spectrum fumigant-like compound methyl isothiocyanate after coming into contact with the wet soil (Matthiessen and Shackleton, 2005). Methyl isothiocyanate (not present in brassicas) is the only isothiocyanate which is used as a pesticide on a commercial basis but other isothiocyanates have also become the subject of interest due to their biofumigation properties (Angus et al., 1994). However, in a cross-enhancement study, Warton et al. (2003) suggested that in soil suffering from enhanced biodegradation of methyl isothiocyanate, biofumigation using isothiocyanate-producing Brassica plants is unlikely to be an effective alternative.

Methyl bromide was used as a soil fumigant with wide-spectrum potential to control soil-borne pests and diseases, nematodes and weeds for economically important crops such as tomato, strawberries and cucurbits (Ploeg, 2008). The phasing out of the ozonedepleting soil fumigant methyl bromide and a higher interest in developing safe and economically viable insect pest management strategies has led to the initiation of alternative approaches. One of these approaches was the process of biofumigation by using macerated Brassica tissues as biofumigant agents and as cover crops (Noble et al., 2002; Matthiessen and Shackleton, 2005; Monfort et al., 2007). The biofumigation process potentially fulfils the requirements of a safe and economical pest management strategy and is included as a non-chemical alternative to methyl bromide by the Methyl Bromide Technical Options Committee (MBTOC, 1997), as documented by Ploeg (2008). Novel approaches to disease control, for example incorporating brassicas to biofumigate soil and engineering solutions to disease control has made The Scottish Agricultural College's (SAC) contribution to supporting the potato industry unique (Smith, 2002).

Smith and Kirkegaard (2002) tested the effect of 2-phenylethyl isothiocyanate under in vitro conditions towards a range of fungi, oomycetes and bacteria. Trichoderma spp. was the most tolerant, while Aphanomyces, Gaeumannomyces, Phytophthora and Thielaviopsis were very sensitive to 2-phenylethyl isothiocyanate. Matthiessen and Shackleton (2005) tested four pure isothiocyanates (methyl, 2-propenyl, benzyl and 2-phenylethyl isothiocyanate), in vapour exposure tests for biological activity against the model soil insect whitefringed weevil both in vitro and in the presence of three contrasting soils and under four temperatures. The results indicated that brassicas rich in aliphatic isothiocyanates are more likely to have the potential to exert stronger isothiocyanate-based biofumigation effects than those similarly rich in aromatic isothiocyanates. The potential of tissue amendments from wild and cultivated Brassica species was assessed to kill the root lesion nematode Pratylenchus neglectus. The amendment of soils with equimolar levels of purified 2-phenylethyl isothiocyanate resulted in comparable levels of nematode mortality, suggesting that 2-phenylethyl glucosinolate has a role in the suppressive impact of Brassica spp. root tissues (Potter et al., 1998). Cox et al. (2006) used B. juncea cv. Pacific Gold (BSM) to control sting nematodes and observed that BSM seed meal with irrigation provided 92% control, while non-irrigated BSM provided 99.5% control. Monfort et al. (2007) evaluated the potential of Brassica species as an alternative control measure for root-knot nematode (*M. incognita*) by using them as green manure amendments prior to

planting. The results of their trial, although variable, showed some promise of Brassica spp. as a biological control option in vegetable production in Georgia. In another study the effects of soil amendment with rapeseed meal from B. napus cv. Dwarf Essex (high glucosinolate concentrations) and Stonewall (low glucosinolate concentrations) on the biological control activity of Trichoderma harzianum towards Sclerotinia sclerotiorum and Aphanomyces euteiches, Dandurand et al. (2000) demonstrated that both Dwarf Essex and Stonewall meals inhibited colonisation of S. sclerotiorum in soil by T. harzianum, from 100% to 0% and 8%, respectively. A. euteiches was significantly reduced by T. harzianum alone (100%), by amendment with Dwarf Essex meal alone (77%), and by T. harzianum in combination with Dwarf Essex meal (100%). Klingen et al. (2002a) analysed the effect of brassicaceous plants on the survival and infectivity of insect pathogenic fungi and their in vitro studies showed that 100 ppm of 2-phenylethyl isothiocyanate completely inhibited growth of Metarhiziumanisopliae and Tolypocladium cylindrosporum. In a study of oilseed rape B. napus extracts in the laboratory for toxicity against adults of Sitophilus oryzae and Rhizopertha dominica at different concentrations, data showed that the surface treatment of wheat seeds with acetone or petroleum ether extracts of B. napus proved to be very effective as they gave 51.8 and 45.0% mortality among S. oryzae adults at 4.0% conc. level, respectively (Salem et al., 2007). While all tested extracts of B. napus showed various toxicities to R. dominica adults, acetone extract was found to have to have the highest effect as it gave 92.4% mortality at 4.0% concentration level, followed by the diethyl ether extract (90%). In a study by Noble et al. (2002), larvae of masked chafer beetles (Cyclocephala spp.) were placed in soil amended with B. juncea (PI 458934) tissue. Allyl isothiocyanate levels were observed to be positively correlated with larval mortality, with the 8% B. juncea treatment resulting in 100% larval mortality with an average allyl isothiocyanate concentration of 11.4 mg  $L^{-1}$  of soil atmosphere.

# 4.2 Brassicas as Cover Crops

"Cover crops slow erosion, improve soil, smother weeds, enhance nutrient and moisture availability, help to control many pests and bring a host of other benefits to farms, and simultaneously can reduce costs, increase profits, and even create new sources of income" (Clark, 2007).

Brassicaceae cover crops are considered to play a role in controlling nematodes, fungi, weeds and diseases by releasing chemical compounds from decomposing residue with promising but variable results among different species, varieties, geographical locations and differences due to the planting dates (Clark, 2007). B. hirta, syn. S. alba, B. juncea, Raphanus sativus and B. napus have been used as cover crops with varying success (Boydston, 2004). Mustard cover crops have been used in Europe and the Pacific Northwest to suppress soil-borne nematodes, pathogens and weeds. Brassica crops such as B. juncea have been reported to improve root health in a subsequent cash crop, such as potatoes, grown after a green incorporated Brassica cover crop (Snapp et al., 2006). Brassica cover crop incorporation as a green manure suppresses some pathogens, including Verticillium in potato; Pythium, Fusarium and Rhizoctonia root rots in beans; Pythium in lettuce; pink rot in onion; Aphanomyces, Pythium, Rhizoctonia and Fusarium root rot in peas; and cavity spot and Fusarium in carrot (summarised by Sanders, 2005 as cited by Snapp et al., 2006). B. juncea, having high levels of glucosinolates along with fungal inhibition properties, was the most effective Brassica crop for reducing powdery scab and common scab disease problems in the field trials conducted (Larkin and Griffin, 2007).

## 4.3 Brassicas as Trap Crops

The practice of trap cropping, i.e. using a sacrificial resource for the pest to attack, in order to protect a valued resource, has been known for centuries (Foster and Harris, 1997). The use of *Brassica* as trap crops has gained interest in recent years. In order to decrease loss to the main crop from insects, trap crops are deployed to attract or catch targeted insects (Shelton and Badenes-Perez, 2006). The trap crop, bearing highly attractive host plants preferred by pests, is planted close to the main crop to protect it from pests. Being attractive, the trap crop captures pests and concentrates them on itself, where they get destroyed before reaching the main crop (Hokkanen, 1991). Recently, the

'push-pull strategy' based on an attractive trap crop has been developed to protect oilseed rape (*B. napus*) from its specialist pests (as reviewed by Cook et al., 2007a). The push-pull strategy "involves the behavioural manipulation of insect pests and their natural enemies via the integration of stimuli that act to make the protected resource unattractive or unsuitable to the pests (push) while luring them toward an attractive source (pull) from where the pests are subsequently removed" (Cook et al., 2007a). The term push-pull was first accepted in Australia in 1987, as a strategy for insect pest management (IPM) (Pyke et al., 1987 as cited by Cook et al., 2007a).

In order to protect spring-sown oilseed rape from two major inflorescence pests, the pollen beetle (*M. aeneus*) and the seed weevil (*C. assimilis*), Cook et al. (2006) followed a strategy of using Brassica as a trap crop. The strategy comprised Starlight, an oilseed rape cultivar with relatively low proportions of alkenyl glucosinolates in the leaves and thereby releasing lower levels of attractive isothiocyanates than conventional cultivars as the main crop, and turnip rape as a trap crop. Turnip rape showed good potential as a trap crop for oilseed rape pests, particularly the pollen beetle, as its odour was more attractive to pests than that of oilseed rape (Cook et al., 2006). A study by Barari et al. (2005) examining turnip rape as a trap crop to reduce oilseed rape infestation, and the effects of insecticide treatment on pest incidence and larval parasitism, showed that the turnip rape trap crop borders reduced P. chrysocephala but not C. pallidactylus infestation of oilseed rape plots. Treatment of the trap crop with insecticide had little effect on either pest or parasitoid incidence in the oilseed rape.

Understanding the mechanisms underlying hostplant preferences of herbivorous pests can lead to improved effectiveness and reliability of the trap crop. With this perspective, Cook et al. (2007b) investigated the behavioural and chemical ecology underlying the success of turnip rape trap crops in protecting oilseed rape from the pollen beetle (*M. aeneus*), which feeds in the flowers and lays its eggs in the buds. Phenylacetaldehyde and  $(E,E) - \alpha$ -farnesene were found to be present in air entrainment samples of both plant species at the flowering growth stage, but only in those of *B. rapa* at the bud stage. The former two compounds were behaviourally active in olfactometer tests. These compounds were suggested to be involved in host location by *M. aeneus* and, at least partially, responsible for the attractiveness of turnip rape and its success as a trap crop to protect oilseed rape from this pest.

Shelton et al. (2008) observed that several types of trap crops have been recommended to control diamondback moth, including *B*. oleracea and *B*. juncea. However, the results varied as populations of P. xylostella develop on these trap crops and spill over to the cash crop. In order to overcome this problem, Shelton et al. (2008) sought to develop "dead-end" trap crops that were more attractive for oviposition than the cash crop and on which P. xylostella larvae cannot survive. Transgenic collard and Indian mustard lines expressing the Bt gene (cry1C) were produced to be used as a "dead-end" trap crop for P. xylostella. The use of Bt Indian mustard as a trap crop significantly reduced the number of larvae that appeared on a cabbage cash crop, compared with using a non-Bt Indian mustard trap crop.

# 4.4 Plant Breeding Towards Insect Resistance in Brassica Crops

Plant breeding has undergone major changes throughout the past 30 years and moved from the traditional phenotype-based via phenotype-protein-based breeding to the utilisation of so-called molecular markers. The development of PCR techniques has given rise to marker technologies and the generation of detailed molecular maps for selection purposes and breeding programmes for important crop plants, also including crucifers. Additionally, the utilisation of model plants, and in particular A. thaliana, has opened up for comparative investigations, not least since Arabidopsis is a close relative to Brassica species, and thus, facilitated and pushed efforts toward molecular breeding (Snowdon and Friedt, 2004) and state-of-the-art metabolomics-assisted breeding (Fernie and Schauer, 2008). Recent approaches for potential genotype selection and cultivar development in the Brassicaceae (Snowdon, 2007; Duran et al., 2009; Nicolas et al., 2009; Riaño-Pachón et al., 2009) include:

- re-synthesis of *Brassica* species through somatic or interspecific hybridisation;
- genetic markers, mapping and identification of quantitative trait loci (QTLs);
- plant model (*Arabidopsis*)-based breeding through intergenome knowledge transfer;

- haploid techniques based on anther and microspore cultures;
- 'omics technologies for global elucidation of genetic, protein and/or metabolite information.

At least two or several of the described approaches are normally applied at the same time, in order to efficiently improve and accelerate the selection process. General breeding goals in *Brassica* crops comprise quite different traits ranging from morphology and yield, oil content and quality, male sterility, and abiotic stress tolerance, to pathogen and insect resistance (Leckband et al., 2002; Snowdon and Friedt, 2004; Sarfraz et al., 2006).

Nevertheless, today's situation in Brassica breeding with regard to insect resistance traits has to be informed on the background of gene modification (GM) of crop plants, on the one hand. Due to successful introduction of genes coding for Bt toxins against insect pests in important crops such as cotton, maize and potato (AGBIOS, 2009), R&D activities in GM Brassica species toward insect resistance traits have been forced in recent years. In cabbage and oilseed rape, the expression of Bt toxins (Jin et al., 2000; Liu et al., 2008), harmful enzymes (Wang et al., 2005; Mulligan et al., 2006) and the regulation of insect gene expression through RNA interference (RNAi) (Baum et al., 2007) have been addressed. However, insect-resistant GM Brassica crops are not commercially available so far. On the other hand, biological premises and limitations challenge the breeding efforts toward insect resistance. Evolutionary-evolved plant responses upon feeding insect species have resulted in specific adapted and partly, multitrophic defence mechanisms, depending on whether generalists or specialists are involved. Thus, insect-plant relationships rely on a complex interplay of factors related to attraction, recognition and oviposition, which makes breeding efforts difficult since several traits have to be taken into account. Rather few examples on breeding Brassicaceae toward insect resistance traits exist. Both the (1) biochemical basis of resistance (glucosinolates; host-plant volatiles; other secondary metabolites) and/or the (2) morphological basis of resistance (leaf colour, size and position; epicuticular waxes; trichome density) might be addressed as reviewed by (Sarfraz et al., 2006). In the case of the glucosinolates and derivatives, intensive research in the past decades has led to new knowledge about both

the potential toxicity of single biochemical compounds and potential induced resistance and thus, opened up new breeding purposes in the brassicas in general (Lou et al., 2008; Hopkins et al., 2009). Although glucosinolates, and specifically their breakdown products, have been shown to play a major role in defence mechanisms against generalist and specialist insects (stressed in Sects. 2 and 3), such interactions have to be studied on a case-by-case basis for a purposeful plant genotype selection, as pointed out for the diamondback moth in oilseed rape (Sarfraz et al., 2006, 2007). Promising biochemical traits other than glucosinolates have been pointed out by Silverstein and co-authors regarding defensins and smaller cysteine-rich peptides for the establishment of constitutive resistance against insect pests (Silverstein et al., 2005, 2007). Trait selections in less important species, e.g. kale (B. oleracea var. acephala) and nabicol (B. napus var. pabularia) (Rodriguez et al., 2005; Picoaga et al., 2003) have shown that the morphological characteristics (glossy leaves) are connected to resistance toward lepidopterous species. In the case of generalist insects, e.g. the cabbage maggot, cross-species approaches using different crucifers might help to identify sources and mechanisms of resistance and thus, to select suitable plant genotypes for hybridisation breeding (Jyoti et al., 2001).

To what extent insect resistance as a newly introduced trait interferes in multitrophic relationships between the host plant, insect pests and their parasitoids has specifically been studied in gene-modified *B. napus* expressing the *Bt* toxin, with negligible effects on beneficial species (Schuler et al., 2001, 2004; Ferry et al., 2006; Mulligan et al., 2006). Recent studies on the glucosinolate content and composition of host plants with differing insect resistance potential have shown that changed plant chemistry possibly impairs parasitoid fitness, underscoring how fine-tuned evolutionary-developed plant herbivore-parasitoid interactions are (Gols et al., 2008); Bukovinszky et al., 2009; Hopkins et al., 2009).

## **5** Conclusions and Future Directions

This review binds together broadly scattered literature on the chemical ecology of *Brassica* plants towards different insects, multiple defence mechanisms of these plants, their role towards the insect-pest complex that attacks brassicas, their potential in insect-pest management and plant breeding towards insect resistance in Brassica. Several chemical ecology studies on different *Brassica* plants and insects attacking brassicas emphasise the importance of glucosinolates, breakdown products (isothiocyanates and nitriles), volatile compounds (terpenes, green leaf volatiles, aromatic compounds) and phytoalexins as key components of plant defence against insects. In addition, these studies show that the Brassica crops exhibiting multiple defence responses in response to different insects along with insect behaviour represent a complex system. In order to understand this complex system further and to identify differential responses triggered by different insects (herbivores, parasitoids/predators) as well as the underlying regulatory networks and signalling pathways, there is a need to perform gene expression studies on brassicas. Due to the availability of the A. thaliana genome, a lot of transcriptional/ecogenomics studies have been performed on Arabidopsis in recent years. Using a full genome microarray platform for studying early transcriptional responses in A. thaliana against B. brassicae infestation, Kuśnierczyk et al. (2008) found strong indications that camalexin is a hitherto unknown insecticide. This finding was further supported by the analysis of camalexin induction and aphid fecundity experiments.

Since Arabidopsis and Brassicas belong to the same family, Arabidopsis microarrays have been employed for the transcriptional studies in Brassica spp. (Dong et al., 2004; Carlsson et al., 2007; Hudson et al., 2007), and can also be used to perform ecogenomics studies to analyse insect infestation responses in Brassica crops. Moreover, with the availability of Brassica microarrays, it is now possible to perform more complete global transcription profiling studies of insect infestation of Brassica plants, providing another gateway towards insect pest management. Future studies could also aim at manipulating plant secondary metabolites such as camalexin, glucosinolates and their breakdown products, volatile compounds, plant allelochemicals or other semiochemicals in order to control pests. This can be applicable through the production of transgenics, through a combination of breeding methods and tissue culture techniques, and through the exploitation of wild Brassica germplasm.

We end this review with a hope that future studies might also discover other aspects of plant-insect interactions, their chemistries in combination with ecogenomics studies, and utilisation of brassicas towards insect-pest management, and will provide us with even more information and clues to understand the plantinsect world and its mysterious trophic interactions. This field of research has gained more and more attention in the past decade and attracted a huge number of scientists, who have spent years exploring the plantinsect world, and have already provided us with excellent information, sources and some natural solutions to overcome insect-pest problems.

Acknowledgements We are grateful to Ingeborg Klingen, Ralph Kissen, Carolyn Baggerud and Sankalp Ahuja for critical reading of the article and providing constructive comments. We further thank three anonymous reviewers for their suggestions. The authors thank Drs. Jean-Marie Bossennec (France), Ian Kimber and Ben Smart (UK), Paolo Mazzei (Italy), and Anna Kusnierczyk (from our group) for providing insect pictures or giving permission to use insect images from homepages: (http://www.inra.fr/hyppz/pests.htm), (http:// ukmoths.org.uk/show.php) and (http://www.leps.it/index.htm) to make Fig. 3. This work was supported by the grants No. 185173 (Plant innate immunity), No. 184146 (A systems biology approach for modelling of plant signalling and host defence) and No. 175691 (NKJ 126 - Genomic analyses and modelling of plant secondary metabolism for development of oilseed rape with improved pest resistance) from the Norwegian Research Council (RCN).

## References

- Acheampong S., Stark J.D. (2004) Can reduced rates of pymetrozine and natural enemies control the cabbage aphid, *Brevicoryne brassicae* (Homoptera: Aphididae), on broccoli, Int. J. Pest Manage. 50, 275–279.
- Agarwala B.K., Datta N. (1999) Life history response of the mustard aphid *Lipaphis erysimi* to phenological changes in its host, J. Bioscience. 24, 224–231.
- AGBIOS (2009) GM database, The AGBIOS Company, Ontario, Canada, http://www.agbios.com/dbase.php.
- Agbogba B., Powell W. (2007) Effect of the presence of a nonhost herbivore on the response of the aphid parasitoid *Diaeretiella rapae* to host-infested cabbage plants, J. Chem. Ecol. 33, 2229–2235.
- Agelopoulos N.G., Keller M.A. (1994a) Plant-natural enemy association in the tritrophic system, *Cotesia rubecula-Pieris rapae*-brassiceae (cruciferae): I. Sources of infochemicals, J. Chem. Ecol. 20, 1725–1734.
- Agelopoulos N.G., Keller M.A. (1994b) Plant-natural enemy association in the tritrophic system *Cotesia rubecula-Pieris rapae*-brassicaceae (cruciferae): II. Preference of *C. rubecula* for landing and searching, J. Chem. Ecol. 20, 1735–1748.
- Agelopoulos N.G., Keller M.A. (1994c) Plant-natural enemy association in the tritrophic system, *Cotesia rubecula-Pieris*

*rapae*-brassicaceae (cruciferae) III: Collection and identification of plant and frass volatiles, J. Chem. Ecol. 20, 1955–1967.

- Agrawal A.A. (2000) Mechanisms, ecological consequences and agricultural implications of tritrophic interactions, Curr. Opin. Plant Biol. 3, 329–335.
- Agrawal A.A., Kurashige N.S. (2003) A role for isothiocyanates in plant resistance against the specialist herbivore *Pieris rapae*, J. Chem. Ecol. 29, 1403–1415.
- Agrawal A.A., Laforsch C., Tollrian R. (1999) Transgenerational induction of defences in animals and plants, Nature 401, 60–63.
- AgroAtlas (2009) Interactive Agricultural Ecological Atlas of Russia and Neighbouring Countries. Economic Plants and their Diseases, Pests and Weeds, Russia, http://www. agroatlas.ru.
- Angus J.F., Gardner P.A., Kirkegaard J.A., Desmarchelier J.M. (1994) Biofumigation: isothiocyanates released from *Brassica* roots inhibit growth of the take-all fungus, Plant Soil 162, 107–112.
- Arab A., Bento J.M.S. (2006) Plant volatiles: new perspectives for research in Brazil, Neotrop. Entomol. 35, 151–158.
- Aslam M., Ahmad M. (2001) Effectiveness of some insecticides against mustard aphid, *Lipaphis erysimi* (Kalt.) (Aphididae: Homoptera) on three different crops, J. Res. (Science) 12, 19–25.
- Åhman I. (1985) Oviposition behaviour of *Dasineura brassicae* on a high versus a low-quality *Brassica* host, Entomol. Exp. Appl. 39, 247–253.
- Åhman I. (1986) Toxicities of host secondary compounds to eggs of the *Brassica* specialist *Dasineura brassicae*, J. Chem. Ecol. 12, 1481–1488.
- Bajwa W.I., Kogan M. (2002) Compendium of IPM Definitions (CID) – What is IPM and how is it defined in the Worldwide Literature? IPPC Publication No. 998, Integrated Plant Protection Center (IPPC), Oregon State University, Corvallis, OR, 97331, USA.
- Baldridge G.D., O'Neill N.R., Samac D.A. (1998) Alfalfa (Medicago sativa L.) resistance to the root-lesion nematode, *Pratylenchus penetrans*: defense-response gene mRNA and isoflavonoid phytoalexin level in roots, Plant Mol. Biol. 38, 999–1010.
- Baldwin I.T., Kessler A., Halitschke R. (2002) Volatile signaling in plant-plant-herbivore interactions: what is real, Curr. Opin. Plant Biol. 5, 351–354.
- Barari H., Cook S.M., Clark S.J., Williams I.H. (2005) Effect of a turnip rape (*Brassica rapa*) trap crop on stem-mining pests and their parasitoids in winter oilseed rape (*Brassica napus*), BioControl 50, 69–86.
- Barker A., Molotsane R., Müller C., Schaffner U., Städler E. (2006) Chemosensory and behavioural responses of the turnip sawfly, *Athalia rosae*, to glucosinolates and isothiocyanates, Chemoecology 16, 209–218.
- Barron A.B. (2001) The life and death of Hopkins' hostselection principle, J. Insect Behav. 14, 725–737.
- Bartlet E., Williams I.H. (1991) Factors restricting the feeding of the cabbage stem flea beetle (*Psylliodes chrysocephala*), Entomol. Exp. Appl. 60, 233–238.
- Bartlet E., Blight M., Hick A., Williams I.H. (1993) The responses of the cabbage seed weevil (*Ceutorhynchus assimilis*) to the odour of oilseed rape (*Brassica napus*) and

to some volatile isothiocyanates, Entomol. Exp. Appl. 68, 295–302.

- Bartlet E., Mithen R., Clark S.J. (1996) Feeding of the cabbage stem flea beetle *Psylliodes chrysocephala* on high and low glucosinolate cultivars of oilseed rape, Entomol. Exp. Appl. 80, 87–89.
- Bartlet E., Blight M., Lane P., Williams I.H. (1997) The responses of the cabbage seed weevil *Ceutorhynchus assimilis* to volatile compounds from oilseed rape in a linear track olfactometer, Entomol. Exp. Appl. 85, 257–262.
- Bartlet E., Kiddle G., Williams I. (1999) Wound-induced increases in the glucosinolate content of oilseed rape and their effect on subsequent herbivory by a crucifer specialist, Entomol. Exp. Appl. 91, 163–167.
- Bartlet E., Blight M., Pickett J., Smart L., Turner G., Woodcock C. (2004) Orientation and feeding responses of the pollen beetle *Meligethes aeneus* to Candytuft *Iberis amara*, J. Chem. Ecol. 30, 913–925.
- Baum J.A., Bogaert T., Clinton W., Heck G.R., Feldmann P., Ilagan O., Johnson S., Plaetinck G., Munyikwa T., Pleau M., Vaughn T., Roberts J. (2007) Control of coleopteran insect pests through RNA interference, Nat. Biotechnol. 25, 1322–1326.
- Baur R., Birch A.N.E., Hopkins R.J., Griffiths D.W., Simmonds M.S.J., Städler E. (1996) Oviposition and chemosensory stimulation of the root flies *Delia radicum* and *D. floralis* in response to plants and leaf surface extracts from resistant and susceptible *Brassica* genotypes, Entomol. Exp. Appl. 78, 61–75.
- Baur R., Städler E., Monde K., Takasugi M. (1998) Phytoalexins from *Brassica* (Cruciferae) as oviposition stimulants for the cabbage root fly, *Delia radicum*, Chemoecology 8, 163–168.
- Behan M., Schoonhoven L.M. (1978) Chemoreception of an oviposition deterrent associated with eggs in *Pieris brassicae*, Entomol. Exp. Appl. 24, 163–179.
- Bellostas N., Sørensen A.D., Sørensen J.C., Sørensen H., Sørensen M.D., Gupta S.K., Kader J.C. (2007) Genetic variation and metabolism of glucosinolates, Adv. Bot. Res. 45, 369–415.
- Bergström G., Rothschild M., Groth I., Crighton C. (1994) Oviposition by butterflies on young leaves: Investigation of leaf volatiles, Chemoecology 5, 147–158.
- Bezemer T., Van Dam N.M. (2005) Linking aboveground and belowground interactions via induced plant defenses, Trends Ecol. Evol. 20, 617–624.
- Birch A.N.E., Griffiths D.W., Hopkins R.J., Macfarlane Smith W.H., McKinlay R.G. (1992) Glucosinolate responses of swede, kale, forage and oilseed rape to root damage by turnip root fly (*Delia floralis*) larvae, J. Sci. Food Agr. 60, 1–9.
- Birkett M.A., Campbell C.A.M., Chamberlain K., Guerrieri E., Hick A.J., Martin J.L., Matthes M., Napier J.A., Pettersson J., Pickett J.A., Poppy G.M., Pow E.M., Pye B.J., Smart L.E., Wadhams G.H., Wadhams L.J., Woodcock C.M. (2000) New roles for *cis*-jasmone as an insect semiochemical and in plant defence, Proc. Natl Acad. Sci. (USA) 97, 9329–9334.
- Blaakmeer A. (1994) Infochemicals in a tritrophic system: interactions between *Brassica*, *Pieris* and *Cotesia*. Dissertation, No. 1802, Wageningen UR Library Catalogue.
- Blaakmeer A., Geervliet J., Van Loon J., Posthumus M., Van Beek T., De Groot A. (1994) Comparative headspace analysis of cabbage plants damaged by two species of *Pieris*

caterpillars: consequences for in-flight host location by *Cotesia* parasitoids, Entomol. Exp. Appl. 73, 175–182.

- Blande J., Pickett J., Poppy G. (2007) A comparison of semiochemically mediated interactions involving specialist and generalist *Brassica*-feeding aphids and the braconid parasitoid *Diaeretiella rapae*, J. Chem. Ecol. 33, 767–779.
- Blight M.M., Smart L.E. (1999) Influence of visual cues and isothiocyanate lures on capture of the pollen beetle, *Meligethes aeneus* in field traps, J. Chem. Ecol. 25, 1501–1516.
- Blight M.M., Pickett J.A., Wadhams L.J., Woodcock C.M. (1995) Antennal perception of oilseed rape, *Brassica napus* (Brassicaceae), volatiles by the cabbage seed weevil *Ceutorhynchus assimilis* (Coleoptera, Curculionidae), J. Chem. Ecol. 21, 1649–1664.
- Blossey B., Hunt-Joshi T.R. (2003) Belowground by insects: Influence on plants and aboveground herbivores, Annu. Rev. Entomol. 48, 521–547.
- Bodnaryk R.P. (1992) Effects of wounding on glucosinolates in the cotyledons of oilseed rape and mustard, Phytochemistry 31, 2671–2677.
- Boman H.G. (1995) Peptide antibiotics and their role in innate immunity, Annu. Rev. Immunol. 13, 61–92.
- Bones A.M. (1990) Distribution of β-thioglucosidase activity in intact plants, cell and tissue cultures and regenerant plants of *Brassica napus* L., J. Exp. Bot. 41, 737–744.
- Bones A., Iversen T.H. (1985) Myrosin cells and myrosinase, Isr. J. Bot. 34, 351–375.
- Bones A.M., Rossiter J.T. (1996) The myrosinase-glucosinolate system, its organisation and biochemistry, Physiol. Plantarum 97, 194–208.
- Bones A.M., Rossiter J.T. (2006) The enzymic and chemically induced decomposition of glucosinolates, Phytochemistry 67, 1053–1067.
- Bones A.M., Slupphaug G. (1989) Purification, characterization and partial amino acid sequencing of  $\beta$ -thioglucosidase from rapeseed, J. Plant Physiol. 134, 722–729.
- Bones A.M., Thangstad O.P., Haugen O., Espevik T. (1991) Fate of myrosin cells - Characterization of monoclonal antibodies against myrosinase, J. Exp. Bot. 42, 1541–1549.
- Bones A.M., Visvalingam S., Thangstad O.P. (1994) Sulphate can induce differential expression of thioglucoside glucohydrolases (myrosinases), Planta 193, 558–566.
- Borregaard N., Elsbach P., Ganz T., Garred P., Svejgaard A. (2000) Innate immunity: from plants to humans, Immunol. Today 21, 68–70.
- Boydston R.A. (2004) Take cover from the elements *Brassica* cover crops can control weeds and reduce the use of crop protectants in vegetable rotations, American Vegetable Grower, March 2004, pp. 18–19.
- Bridges M., Jones A.M.E., Bones A.M., Hodgson C., Cole R., Bartlet E., Wallsgrove R., Karapapa V.K., Watts N., Rossiter J.T. (2002) Spatial organization of the glucosinolatemyrosinase system in *Brassica* specialist aphids is similar to that of the host plant, Proc. R. Soc. Lond. B 269, 187–191.
- Bromand B. (1990) Diversities in oilsed rape growing within the western palaearctic regional section, IOBC/WPRS Bull. 13, 7–31.
- Brown P.D., Morra M.J. (1997) Control of soil-borne plant pests using glucosinolate-containing plants, Adv. Agron. 61, 67–231.

- Bruinsma M., Van Dam N., Van Loon J., Dicke M. (2007) Jasmonic acid-induced changes in *Brassica oleracea* affect oviposition preference of two specialist herbivores, J. Chem. Ecol. 33, 655–668.
- Bruinsma M., Posthumus M.A., Mumm R., Mueller M.J., Van Loon J., Dicke M. (2009) Jasmonic acid-induced volatiles of *Brassica oleracea* attract parasitoids: effects of time and dose, and comparision with induction by herbivores, J. Exp. Bot., DOI: 10.1093/jxb/erp101.
- Bukovinszky T., Poelman E.H., Gols R., Prekatsakis G., Vet L.E., Harvey J.A., Dicke M. (2009) Consequences of constitutive and induced variation in plant nutritional quality for immune defence of a herbivore against parasitism, Oecologia 160, 299–308.
- Burmeister W., Härtel F., Brandt A., Rossiter J.T., Bones A.M. (2005) The crystal structure at 1.1 Å resolution of myrosinase from the aphid *Brevicoryne brassicea* shows its close similarity with  $\beta$ -glucosidases, Insect Biochem. Molec. 35, 1311–1320.
- Capinera J.L. (2004) Cabbageworm, *Pieris rapae* (Linnaeus) (Lepidoptera: Pieridae), Encyclopedia of Entomology, Kluwer, Dordrecht, The Netherlands, pp. 444–445.
- Capinera J.L. (2008) Cabbage looper, *Trichoplusia ni* (Hübner) (Insecta: Lepidoptera: Noctuidae), EENY-116, Florida Cooperative Extension Service, University of Florida IFAS Extension.
- Cardé R.T., Millar J.G. (2004) Advances in insect chemical ecology, Cambridge Univ. Press, Cambridge, UK.
- Carlsson J., Lagercrantz U., Sundström J., Teixeira R., Wellmer F., Meyerowitz E., Glimelius K. (2007) Microarray analysis reveals altered expression of a large number of nuclear genes in developing cytoplasmic male sterile *Brassica napus* flowers, Plant J. 49, 452–462.
- Cartea M.E., Padilla G., Vilar M., Velasco P. (2009) Incidence of the major *Brassica* pests in northeastern Spain, J. Econ. Entomol. 102, 767–773.
- Chadchawan S, Bishop J., Thangstad O.P., Bones A.M., Mitchell-Olds T., Bradley D. (1993) *Arabidopsis* cDNA sequence encoding myrosinase, Plant Physiol. 103, 671–672.
- Charleston D.S., Kfir R. (2000) The possibility of using Indian mustard, *Brassica juncea*, as a trap crop for the diamondback moth *Plutella xylostella* in South Africa, Crop Prot. 19, 455–460.
- Chattopadhyay C., Agrawal R., Kumar A., Singh Y.P., Roy S.K., Khan S.A., Bhar L.M., Chakravarthy N.V.K., Srivastava A., Patel B.S., Srivastava B., Singh C.P., Mehta S.C. (2005) Forecasting of *Lipaphis erysimi* on oilseed Brassicas in India - a case study, Crop Prot. 24, 1042–1053.
- Chow J.K., Akhtar Y., Isman M.B. (2005) The effects of larval experience with a complex plant latex on subsequent feeding and oviposition by the cabbage looper moth: *Trichoplusia ni* (Lepidoptera: Noctuidae), Chemoecology 15, 129–133.
- Cintas N.A., Koike S.T., Bull C.T. (2002) A new pathovar, *Pseudomonas syringae* pv. *alisalensis* pv. *nov.*, proposed for the causal agent of bacterial blight of broccoli and broccoli raab, Plant Dis. 86, 992–998.
- Cirio V. (1971) Reperti sul meccanismo stimolo-riposta nell' ovideposizione del *Dacus oleae* Gmelin (Diptera, Trypetidae), Redia 52, 557–600.
- Clark A. (2007) Managing cover crops profitably (3rd ed.), Sustainable Agricultural Network (SAN), Handbook series

Book 9, A publication of the Sustainable Agriculture Network with funding by the Sustainable Agriculture Research and Education Program of CSREES, U.S. Department of Agriculture. Beltsville, MD, USA.

- Clossais-Bernard N., Larher F. (1991) Physiological role of glucosinolates in *Brassica napus*, Concentration and distribution pattern of glucosinolates among plant organs during a complete life cycle, J. Sci. Food Agr. 56, 25–38.
- Cole R.A. (1994) Locating a resistance mechanism to the cabbage aphid in two wild Brassicas, Entomol. Exp. Appl. 71, 23–31.
- Cole R.A. (1997) Comparison of feeding behaviour of two Brassica pests Brevicoryne brassicae and Myzus persicae on wild and cultivated Brassica species, Entomol. Exp. Appl. 85, 135–143.
- Cook S.M., Smart L.E., Martin J.L., Murray D.A., Watts N.P., Williams I.H. (2006) Exploitation of host plant preferences in pest management strategies for oilseed rape (*Brassica napus*), Entomol. Exp. Appl. 119, 221–229.
- Cook S.M., Khan Z.R., Pickett J.A. (2007a) The use of push-pull strategies in integrated pest management, Annu. Rev. Plant Biol. 52, 375–400.
- Cook S.M., Rasmussen H., Birkett M., Murray D., Pye B., Watts N., Williams I. (2007b) Behavioural and chemical ecology underlying the success of turnip rape (*Brassica* rapa) trap crops in protecting oilseed rape (*Brassica napus*) from the pollen beetle (*Meligethes aeneus*), Arthropod-Plant Interact. 1, 57–67.
- Cooper R.M., Resende M.L.V., Flood J., Rowan M.G., Beale M.H., Potter U. (1996) Detection and cellular localization of elemental sulphur in disease resistant genotypes of *Theobroma cacao*, Nature 379, 159–162.
- Cox C.J., Lambert B., McCarty L.B., Toler, J.E., Lewis, S.A., Bruce Martin S. (2006) Suppressing sting nematodes with *Brassica* sp., poinsettia, and spotted spurge extracts, Agron. J. 98, 962–967.
- Damodaram T., Hegde D.M. (2002) Oilseeds Situation: A Statistical Compendium 2002, Directorate of Oilseeds Research, Hyderabad 500030, India, p. 471.
- Dandurand L.M., Mosher R.D., Knudsen G.R. (2000) Combined effects of *Brassica napus* seed meal and *Trichoderma harzianum* on two soilborne plant pathogens, Can. J. Microbiol. 46, 1051–1057.
- David W.A.L., Gardiner B.O.C. (1966) The effect of sinigrin on the feeding of *Pieris brassicae* L. larvae transferred from various diets, Entomol. Exp. Appl. 9, 95–98.
- Dawson G.W., Griffiths D.C., Pickett J.A., Wadhams L.J., Woodcock C.M. (1987) Plant-derived synergists of alarm pheromone from turnip aphid, *Lipaphis* (Hyadaphis) *erysimi* (Homoptera, Aphididae), J. Chem. Ecol. 13, 1663–1671.
- Dawson G.W., Griffiths D.C., Merritt L.A. Mudd A., Pickett J.A., Wadhams L.J., Woodcock C.M. (1990) Aphid semiochemicals - A review, and recent advances on the sex pheromone, J. Chem. Ecol. 16, 3019–3030.
- De Boer J., Hordijk C., Posthumus M., Dicke M. (2008) Prey and non-prey arthropods sharing a host plant: Effects on induced volatile emission and predator attraction, J. Chem. Ecol. 34, 281–290.
- De Jong R., Städler E. (1999) The influence of odour on the oviposition behaviour of the cabbage root fly, Chemoecology 9, 151–154.

- De Jong R., Maher N., Patrian B. Städler E., Winkler T. (2000) Rutabaga roots, a rich source of oviposition stimulants for the cabbage root fly, Chemoecology 10, 205–209.
- Dicke M. (1999) Are herbivore-induced plant volatiles reliable indicators of herbivore identity to foraging carnivorous arthropods, Entomol. Exp. Appl. 91, 131–142.
- Dicke M., Sabelis M.W. (1989) Does it pay plants to advertise for bodyguards? Towards a cost-benefit analysis of induced synonome production, SPB Academic Publishing, The Hague, The Netherlands.
- Dicke M., Sabelis M.W. (1992) Costs and benefts of chemical information conveyance: proximate and ultimate factors, in: Roitberg B.D., Isman M.B. (Eds.), Insect Chemical ecology, An evolutionary approach, Chapman and Hall, New York, USA, pp. 122–155.
- Dicke M., Van Loon J.J.A. (2000) Multitrophic effects of herbivore-induced plant volatiles in an evolutionary context, Entomol. Exp. Appl. 97, 237–249.
- Dicke M., Hilker M. (2003) Induced plant defences: from molecular biology to evolutionary ecology, Basic Appl. Ecol. 4, 3–14.
- Dicke M., Sabelis M.W., Takabayashi J., Bruin J., Posthumus M.A. (1990) Plant strategies of manipulating predator prey interactions through allelochemicals: prospects for application in pest control, J. Chem. Ecol. 16, 3091–3118.
- Dicke M., De Boer J.G., Hofte M., Rocha-Granados M.C. (2003) Mixed blends of herbivore-induced plant volatiles and foraging success of carnivorous arthropods, Oikos 101, 38–48.
- Dicke M., Van Loon J.J.A., De Jong P.W. (2004) Ecology: enhanced: ecogenomics benefits community ecology, Science 305, 618–619.
- Dixon R.A. (2001) Natural products and plant disease resistance, Nature 411, 843–847.
- Dong J.Z., Keller W.A., Yan W., Georges F. (2004) Gene expression at early stages of *Brassica napus* seed development as revealed by transcript profiling of seed-abundant cDNAs, Planta 218, 483–491.
- Du Toit L.J. (2007) Crop profile for cabbage seed in Washington, Washington State University Extension bulletin, MISC0358E.
- Dubuis P.H. (2004) Effect of sulfur deficiency on the resistance of oilseed rape to fungal pathogens and expression profiling of the glutathione S- transferase family of Arabidopsis thaliana, Doctoral Thesis, Institute of Plant Biology, University of Fribourg, Switzerland.
- Dubuis P.H., Marazzi C., Stadler E., Mauch F. (2005) Sulphur deficiency causes a reduction in antimicrobial potential and leads to increased disease susceptibility of oilseed rape, J. Phytopathol. 153, 27–36.
- Duran C., Edwards D., Batley J. (2009) Genetic maps and the use of synteny, Methods Mol. Biol. 513, 41–55.
- Dutta I., Majumder P., Saha P., Ray K., Das S. (2005) Constitutive and phloem specific expression of *Allium sativum* leaf agglutinin (ASAL) to engineer aphid (*Lipaphis erysimi*) resistance in transgenic Indian mustard (*Brassica juncea*), Plant Sci. 169, 996–1007.
- Ekbom B. (1995) Insect pests of *Brassica* oilseeds, in: Kimber D., McGregor I. (Eds.), *Brassica* Oilseeds- Production and Utilisation, CAB International, UK.

- Ekbom B., Borg A. (1996) Pollen beetle (*Meligethes aeneus*) oviposition and feeding preference on different host plant species, Entomol. Exp. Appl. 78, 291–299.
- Ekbom B., Ferdinand V. (2003) Field oviposition rates and egg load dynamics of pollen beetles (*Meligethes aeneus* Fab.) (Colepotera: Nitidulidae), Agr. Forest Entomol. 5, 247–252.
- Ester A., De Putter H., Van Bilsen J.G.P.M. (2003) Filmcoating the seed of cabbage (*Brassica oleracea* L. convar. *Capitata* L.) and cauliflower (*Brassica oleracea* L. var. *Botrytis* L.) with imidacloprid and spinosad to control insect pests, Crop Prot. 22, 761–768.
- Evans K.A., Allen-Williams L.J. (1992) Electroantennogram responses of the cabbage seed weevil, *Ceutorhynchus assimilis* to oilseed rape *Brassica napus* ssp. *oleifera*, volatiles, J. Chem. Ecol. 18, 1641–1659.
- Evans K.A., Allen-Williams L.J. (1994) Laboratory and field response of the pollen beetle, *Meligethes aeneus*, to the odour of oilseed rape, Physiol. Entomol. 19, 285–290.
- Evans K.A., Allen-Williams L.J. (1998) Response of cabbage seed weevil (*Ceutorhynchus assimilis*) to baits of extracted and synthetic host-plant odor, J. Chem. Ecol. 24, 2101–2114.
- Fahey J.W., Zalcmann A.T., Talalay P. (2001) The chemical diversity and distribution of glucosinolates and isothiocyanates among plants, Phytochemistry 56, 5–51.
- Falk A., Taipalensuu J., Ek B., Lenman M., Rask L. (1995) Characterization of rapeseed myrosinase-binding protein, Planta 195, 387–395.
- FAOSTAT (2009) Publically available databases from: Food and Agricultural Organization of the United Nations (FAO), http://faostat.fao.org/.
- Fatouros N.E., Bukovinszkine'Kiss G., Kalkers L.A., Gamborena R.S., Dicke M., Hilker M. (2005a) Ovipositioninduced plant cues: do they arrest *Trichogramma* wasps during host location, Entomol. Exp. Appl. 115, 207–215.
- Fatouros N.E., Van Loon J.J.A., Hordijk K.A., Smid H.M., Dicke M. (2005b) Herbivore-induced plant volatiles mediate in-flight host discrimination by parasitoids, J. Chem. Ecol. 31, 2033–2047.
- Ferguson A.W., Williams I.H. (1991) Deposition and longevity of oviposition-deterring pheromone in the cabbage seed weevil, Physiol. Entomol. 16, 27–33.
- Ferguson A.W., Ziesmann J., Blight M.M., Williams I.H., Wadhams L.J., Clark S.J., Woodcock C.M., Mudd A. (1999a) Perception of oviposition-deterring pheromone by cabbage seed weevil (*Ceutorhynchus assimilis*), J. Chem. Ecol. 25, 1655–1670.
- Ferguson A.W., Solinas M., Ziesmann J., Isidoro N., Williams I.H., Scubla P., Mudd A., Clark S.J., Wadhams L.J. (1999b) Identification of the gland secreting oviposition-deterring pheromone in the cabbage seed weevil, *Ceutorhynchus assimilis* and the mechanism of pheromone deposition, J. Insect Physiol. 45, 687–699.
- Fernie A.R., Schauer N. (2008) Metabolomics-assisted breeding: a viable option for crop improvement? Trends Genet. 25, 39–48.
- Ferry N., Mulligan E.A., Stewart C.N., Tabashnik B.E., Port G.R., Gatehouse A.M. (2006) Prey-mediated effects of transgenic canola on a beneficial, non-target, carabid beetle, Transgenic Res. 15, 501–514.
- Font R., Del Rio-Celestino M., Fernandez J.M., De Haro A. (2003) Acid detergent fiber analysis in oilseed brassicas

by near-infrared spectroscopy, J. Agr. Food Chem. 51, 2917–2922.

- Font R., Del Rio-Celestino M., Rosa E., Aires A., De Hardo-Bailon A. (2005) Glucosinolate assessment in *Brassica oleracea* leaves by near-infrared spectroscopy, J. Agr. Sci. 143, 65–73.
- Foo H.L., Grønning L.M., Goodenough L., Bones A.M., Danielsen B.E., Whiting D.A., Rossiter J.T. (2000) Purification and characterisation of epithiospecifier protein from *Brassica napus*: Enzymic intramolecular sulphur addition within alkenyl thiohydroximates derived from alkenyl glucosinolate hydrolysis, FEBS Lett. 486, 243–246.
- Foster S.P., Harris M.O. (1997) Behavioural manipulation methods for insect pest-management, Annu. Rev. Entomol. 42, 123–146.
- Francis F., Lognay G., Wathelet J.P., Haubruge E. (2001) Effects of allelochemicals from first (Brassicaceae) and second (*Myzus persicae* and *Brevicoryne brassicae*) trophic levels on *Adalia bipunctata*, J. Chem. Ecol. 27, 243–256.
- Gange A.C., Brown V.K. (1997) Multitrophic interactions in terrestrial systems, Blackwell Science, Oxford, UK.
- Gardiner J.B., Morra M.J., Eberlein C.V., Brown P.D., Borek V. (1999) Allelochemicals released in soil following incorporation of rapeseed (*Brassica napus*) green manures, J. Agr. Food Chem. 47, 3837–3842.
- Gardner S.M., Dissevelt M., Van Lenteren J.C. (2007) Behavioural adaptations in host finding by *Trichogramma evanescens*: the influence of oviposition experience on response to host contact kairomones, B. Insectol. 60, 23–30.
- Geervliet J., Vet L.E.M., Dicke M. (1994) Volatiles from damaged plants as major cues in long-range host-searching by the specialist parasitoid *Cotesia rubecula*, Entomol. Exp. Appl. 73, 289–297.
- Geervliet J.B.F., Posthumus M.A., Vet L.E.M., Dicke M. (1997) Comparative analysis of headspace volatiles from different caterpillar-infested or uninfested food plants of *Pieris* species, J. Chem. Ecol. 23, 2935–2954.
- Geervliet J.B.F., Vreugdenhil A.I., Dicke M., Vet L.E.M. (1998) Learning to discriminate between infochemicals from different plant-host complexes by the parasitoids *Cotesia glomerata* and *C. rubecula*, Entomol. Exp. Appl. 86, 241–252.
- Girard C., Bonade-Bottino M., Pham-Delegue M.H., Jouanin L. (1998) Two strains of cabbage seed weevil (Coleoptera: Curculionidae) exhibit differential susceptibility to a transgenic oilseed rape expressing oryzacystatin I, J. Insect Physiol. 44, 569–577.
- Golizadeh A.L.I., Kamali K., Fathipour Y. Abbasipour H. (2007) Temperature-dependent development of diamondback moth, *Plutella xylostella* (Lepidoptera: Plutellidae) on two brassicaceous host plants, Insect Sci. 14, 309–316.
- Gols R., Bukovinszky T., Van Dam N., Dicke M., Bullock J., Harvey J. (2008a) Performance of generalist and specialist herbivores and their endoparasitoids differs on cultivated and wild *Brassica* Populations, J. Chem. Ecol. 34, 132–143.
- Gols R., Wagenaar R., Bukovinszky T., Van Dam N.M., Dicke M., Bullock J.M., Harvey J.A. (2008b) Genetic variation in defense chemistry in wild cabbages affects herbivores and their endoparasitoids, Ecology 89, 1616–1626.
- Gols R., Harvey J. (2009) Plant-mediated effects in the Brassicaceae on the performance and behaviour of parasitoids, Phytochem. Rev. 8, 187–206.

- Gomez-Campo C. (1999) Taxonomy, in: Gomez-Campo C. (Ed.), Development in Plant Genetics and Breeding. Elsevier, Amsterdam, The Netherlands, pp. 3–23.
- Görnitz K. (1956) Weitere Untersuchungen über Insekten-Attraktivstoffe aus Cruciferen, Nachrichtenbl. Dtsch. Pflanzenschutzdienst N.F. 10, 137–147.
- Graner G., Hamberg M., Meijer J. (2003) Screening of oxylipins for control of oilseed rape (*Brassica napus*) fungal pathogens, Phytochemistry 63, 89–95.
- Griffiths D.W., Birch A.N.E., Macfarlane-Smith W.H. (1994) Induced changes in the indole glucosinolate content of oilseed and forage rape (*Brassica napus*) plants in response to either turnip-root-fly (*Delia floralis*) larval feeding or artificial root damage, J. Sci. Food Agr. 65, 171–178.
- Gripenberg S., Roslin T. (2007) Up or down in space? Uniting the bottom-up versus top-down paradigm and spatial ecology, Oikos 116, 181–188.
- Guerena M. (2006) Cole crops and other Brassicas: Organic production, ATTRA National Sustainable Agriculture Information Service, NCAT, http://attra.ncat.org.
- Guignard L. (1890) Sur la localization des principes qui fournissent les essences sulfurées des crucifers, C.R. Acad. Hebd. - Seances III, 249–251.
- Gupta P.D., Thorsteinson A.J. (1960) Food plant relationships of the diamond-back moth (*Plutella maculipennis*) (Curt.), Entomol. Exp. Appl. 3, 305–314.
- Halkier B.A., Gershenzon J. (2006) Biology and biochemistry of glucosinolates, Ann. Rev. Plant Biol. 57, 303–333.
- Hamilton A.J., Endersby N.M., Ridland P.M., Zhang J., Neal M. (2005) Effects of cultivar on oviposition preference, larval feeding and development time of diamondback moth, *Plutella xylostella* (L.) (Lepidoptera: Plutellidae), on some *Brassica oleracea* vegetables in Victoria, Aust. J. Entomol. 44, 284–287.
- Hansen L.M. (2004) Economic damage threshold model for pollen beetles (*Meligethes aeneus* F.) in spring oilseed rape (*Brassica napus* L.) crops, Crop Prot. 23, 43–46.
- Harvey J.A., Van Dam N.M., Gols R. (2003) Interactions over four trophic levels: foodplant quality affects development of a hyperparasitoid as mediated through a herbivore and its primary parasitoid, J. Anim. Ecol. 72, 520–531.
- Haukioja E. (1999) Ecology: bite the mother, fight the daughter, Nature 401, 22–23.
- Hawkes C., Coaker T.H. (1979) Factors affecting the behavioural responses of the adult cabbage root fly, *Delia brassicae*, to host plant odour, Entomol. Exp. Appl. 25, 45–58.
- Heil M. (2008) Indirect defence via tritrophic interactions, New Phytol. 178, 41–61.
- Heinricher E. (1884) Über Eiweissstoffe führende Idioblasten bei einigen Cruciferen, Ber. Dtsch. Bot. Ges. II, 463–467.
- Hern A., Edwards-Jones G., McKinlay R.G. (1996) A review of the pre-oviposition behaviour of small cabbage white butterfly *Pieris rapae* (Lepidoptera: Pieridae), Ann. Appl. Biol. 128, 349–371.
- Hilker M., Meiners T. (2002) Induction of plant responses to oviposition and feeding by herbivorous arthropods: a comparison, Entomol. Exp. Appl. 104, 181–192.
- Hilker M., Meiners T. (2006) Early herbivore alert: Insect eggs induce plant defence, J. Chem. Ecol. 32, 1379–1397.
- Hodek I. (1959) The influence of aphid species as food for the ladybirds Coccinella 7-punctata L. and Adalia bipunctata L.,

in: The Ontogeny of Insects, Symposium on the Ontogenetic Development of Insects, Prague. Academic Press, London-New York, pp. 314–316.

- Hokkanen H.M.T. (1991) Trap cropping in pest management, Annu. Rev. Entomol. 36, 119–138.
- Hokkanen H.M.T., Wearing C.H. (1996) Assessing the risk of pest resistance evolution to *Bacillus thuringiensis* engineered into crop plants: a case study of oilseed rape, Field Crop. Res. 45, 171–179.
- Hong C.P., Kwon S.J., Kim J.S., Yang T.J., Park B.S., Lim Y.P. (2008) Progress in understanding and sequencing the genome of *Brassica rapa*, Int. J. Plant Genomics, DOI: 10.1155/2008/582837.
- Hopkins A.D. (1917) A discussion of C.G. Hewitt's paper on "Insect Behaviour", J. Entomol. 10, 92–93.
- Hopkins R.J., Birch A.N.E., Griffiths D.W., Baur R., Städler E., McKinlay R.G. (1997) Leaf surface compounds and oviposition preference of turnip root fly *Delia floralis*: The role of glucosinolate and nonglucosinolate compounds, J. Chem. Ecol. 23, 629–643.
- Hopkins R.J., Griffiths D.W., Birch A.N.E., McKinlay R.G. (1998) Influence of increasing herbivore pressure on modification of glucosinolate content of swedes (*Brassica napus* spp. *rapifera*), J. Chem. Ecol. 24, 2003–2019.
- Hopkins R.J., Wright F., Birch A.N.E., McKinlay R.G. (1999) The decision to reject an oviposition site: sequential analysis of the post-alighting behaviour of *Delia floralis*, Physiol. Entomol. 24, 41–50.
- Hopkins R.J., Van Dam N.M., Van Loon J.J. (2009) Role of glucosinolates in insect-plant relationships and multitrophic interactions, Annu. Rev. Entomol. 54, 57–83.
- Hossain M.A., Maiti M.K., Basu A., Sen S., Ghosh A.K., Sen S.K. (2006) Transgenic expression of onion leaf lectin gene in Indian Mustard offers protection against aphid colonization, Crop Sci. 46, 2022–2032.
- Höglund A.S., Lenman M., Falk A., Rask L. (1991) Distribution of myrosinase in rapeseed tissues, Plant Physiol. 95, 213–221.
- Huang X., Renwick J.A.A. (1993) Differential selection of host plants by two *Pieris* species: the role of oviposition stimulants and deterrents, Entomol. Exp. Appl. 68, 59–69.
- Huang X., Renwick J.A.A. (1994) Relative activities of glucosinolates as oviposition stimulants for *Pieris rapae* and *P. napi oleracea*, J. Chem. Ecol. 20, 1025–1037.
- Hudson M.E., Bruggink T., Chang S.H., Yu W., Han B., Wang X., Toorn P.V.D., Zhu T. (2007) Analysis of gene expression during *Brassica* seed germination using a crossspecific microarray platform, Crop Sci. 47, 96–112.
- Hughes P.R., Renwick J.A.A., Lopez K.D. (1997) New oviposition stimulants for the diamondback moth in cabbage, Entomol. Exp. Appl. 85, 281–283.
- Hurter J., Ramp T., Patrian B., Städler E., Roessingh P., Baur R., De Jong R., Nielsen J.K., Winkler T., Richter W.J., Muller D., Ernst B. (1999) Oviposition stimulants for the cabbage root fly: isolation from cabbage leaves, Phytochemistry 51, 377–382.
- Husebye H., Chadchawan S., Winge P., Thangstad O.P., Bones A.M. (2002) Guard cell- and phloem idioblast-specific expression of thioglucoside glucohydrolase 1 (myrosinase) in *Arabidopsis*, Plant Physiol. 128, 1180–1188.

- Husebye H., Arzt S., Burmeister W.P., Hartel F.V., Brandt A., Rossiter J.T., Bones A.M. (2005) Crystal structure at 1.1 Å resolution of an insect myrosinase from *Brevicoryne brassicae* shows its close relationship to beta-glucosidases, Insect Biochem. Molec. 35, 1311–1320.
- Jin R.G., Liu Y.B., Tabashnik B.E., Borthakur D. (2000) Development of transgenic cabbage (*Brassica oleracea* var. capitata) for insect resistance by Agrobacterium tumefaciensmediated transformation, In Vitro Cell. Dev. B. 36, 231–237.
- Johansen N.S. (1997) Mortality of eggs, larvae and pupae and larval dispersal of the cabbage moth, *Mamestra brassicae*, in white cabbage in South-Eastern Norway, Entomol. Exp. Appl. 83, 347–360.
- Jones A.M.E., Bridges M., Bones A.M., Cole R., Rossiter J.T. (2001) Purification and characterisation of a non-plant myrosinase from the cabbage aphid *Brevicoryne brassicae* (L.), Insect Biochem. Molec. 31, 1–5.
- Jones A.M.E., Winge P., Bones A.M., Cole R., Rossiter J.T. (2002) Characterization and evolution of a myrosinase from the cabbage aphid *Brevicoryne brassicae*, Insect Biochem. Molec. 32, 275–284.
- Jones D.A., Takemoto D. (2004) Plant innate immunity direct and indirect recognition of general and specific pathogenassociated molecules, Curr. Opin. Immunol. 16, 48–62.
- Jyoti J.L., Shelton A.M., Earle E.D. (2001) Identifying sources and mechanisms of resistance in crucifers for control of cabbage maggot (Diptera: Anthomyiidae), J. Econ. Entomol. 94, 942–949.
- Jönsson M., Anderson P. (2007) Emission of oilseed rape volatiles after pollen beetle infestation; behavioural and electrophysiological responses in the parasitoid *Phradis morionellus*, Chemoecology 17, 201–207.
- Jönsson M., Lindkvist A., Anderson P. (2005) Behavioural responses in three ichneumonid pollen beetle parasitoids to volatiles emitted from different phenological stages of oilseed rape, Entomol. Exp. Appl. 115, 363–369.
- Kanrar S., Venkateswari J., Kirti P.B., Chopra, V.L. (2002) Transgenic Indian mustard (*Brassica juncea*) with resistance to the mustard aphid (*Lipaphis erysimi* Kalt.), Plant Cell Rep. 20, 976–981.
- Karban R., Baldwin I.T. (1997) Induced resonses to herbivory, Chicago University Press, Chicago, IL, USA.
- Karimzadeh J., Wright D.J. (2008) Bottom-up cascading effects in a tritrophic system: interactions between plant quality and host-parasitoid immune responses, Ecol. Entomol. 33, 45–52.
- Karowe D.N., Schoonhoven L.M. (1992) Interactions among three trophic levels: the influence of host plant on performance of *Pieris brassicae* and its parasitoid *Cotesia glomerata*, Entomol. Exp. Appl. 62, 241–241.
- Kazachkova N.I. (2007) Genotype analysis and studies of pyrethroid resistance of the oilseed rape (*Brassica napus*) insect pest-pollen beetle (*Meligethes aeneus*), Doctoral thesis, Faculty of Natural Resources and Agricultural Sciences, Department of Plant Biology and Forest Genetics, Swedish University of Agricultural Sciences, Uppsala.
- Kazana E., Pope T.W., Tibbles L., Bridges M., Pickett J.A., Bones A.M., Powell G., Rossiter J.T. (2007) The cabbage aphid: a walking mustard oil bomb, Proc. R. Soc. Lond. B 274, 2271–2277.

- Keiichi H. (1995) Chemical basis of differential oviposition by lepidopterous insects, Arch. Insect Biochem. Physiol. 30, 1–23.
- Kelly P.J., Bones A., Rossiter J.T. (1998) Sub-cellular immunolocalization of the glucosinolate sinigrin in seedlings of *Brassica juncea*, Planta 206, 370–377.
- Kessler A., Baldwin I.T. (2002) Plant responses to insect herbivory: The emerging molecular analysis, Annu. Rev. Plant Biol. 53, 299–328.
- Khattab H. (2007) The defense mechanism of cabbage plant against phloem-sucking aphid (*Brevicoryne brassicae* L.), Aust. J. Basic Appl. Sci. 1, 56–62.
- Kift N.B., Ellis P.R., Tatchell G.M., Pink D.A.C. (2000) The influence of genetic background on resistance to the cabbage aphid (*Brevicoryne brassicae*) in kale (*Brassica oleracea* var. *acephala*), Ann. Appl. Biol. 136, 189–195.
- Kim J.H., Jander G. (2007) *Myzus persicae* (green peach aphid) feeding on *Arabidopsis* induces the formation of a deterrent indole glucosinolate, Plant J. 49, 1008–1019.
- Kirkegaard J.A., Sarwar M. (1998) Biofumigation potential of brassicas I. Variation in glucosinolate profiles of diverse fieldgrown brassicas, Plant Soil 201, 71–89.
- Kirk-Spriggs A.H. (1996) Pollen beetles. Coleoptera: Kateretidae and Nitidulidae: Meligethinae, Handbooks for the identification of British insects, Royal Entomological Society, London, UK.
- Kishimoto K., Matsui K., Ozawa R., Takabayashi J. (2005) Volatile C<sub>6</sub>-aldehydes and *allo*-ocimene activate defense genes and induce resistance against *Botrytis cinerea* in *Arabidopsis thaliana*, Plant Cell Physiol. 46, 1093–1102.
- Kissen R., Rossiter J.T., Bones A.M. (2009) The "mustard oil bomb": not so easy to assemble?! Localization, expression and distribution of the components of the myrosinase enzyme system, Phytochem. Rev. 8, 69–86.
- Klijnstra J.W., Roessingh P. (1986) Perception of the oviposition deterring pheromone by tarsal and abdominal contact chemoreceptors in *Pieris brassicae*, Entomol. Exp. Appl. 40, 71–79.
- Klingen I., Hajek A., Meadow R., Renwick J.A.A. (2002a) Effect of brassicaceous plants on the survival and infectivity of insect pathogenic fungi, BioControl 47, 411–425.
- Klingen I., Meadow R., Aandal T. (2002b) Mortality of *Delia floralis*, *Galleria mellonella* and *Mamestra brassicae* treated with insect pathogenic hyphomycetous fungi, J. Appl. Entomol. 126, 231–237.
- Knodel J.J., Olsen D.L. (2002) Crucifer flea beetle. Biology and integrated pest management in canola, E-1234, NDSU Extension Service, North Dakota State University, USA, http:// www.ag.ndsu.edu/pubs/plantsci/pests/e1234.pdf.
- Knodel J., Ganehiarachchi M. (2008) Bertha armyworm in canola. Biology and integrated pest management in canola, E-1347, NDSU Extension Service, North Dakota State University, USA, http://www.ag.ndsu.edu/pubs/plantsci/pests/ e1347.pdf.
- Koritsas V.M., Lewis J.A., Fenwick G.R. (1991) Glucosinolate responses of oilseed rape, mustard and kale to mechanical wounding and infestation by cabbage stem flea beetle (*Psylliodes chrysocephala*), Ann. Appl. Biol. 118, 209–221.
- Koroleva O.A., Davies A., Deeken R., Thorpe M.R., Tomos A.D., Hedrich R. (2000) Identification of a new

glucosinolate-rich cell type in *Arabidopsis* flower stalk, Plant Physiol. 124, 599–608.

- Kuśnierczyk A., Winge P., Midelfart H., Armbruster W.S., Rossiter J.T., Bones A.M. (2007) Transcriptional responses of *Arabidopsis thaliana* ecotypes with different glucosinolate profiles after attack by polyphagous *Myzus persicae* and oligophagous *Brevicoryne brassicae*, J. Exp. Bot. 58, 2537–2552.
- Kuśnierczyk A., Winge P., Jørstad T.S., Troczyńska J., Rossiter J.T., Bones A.M. (2008) Towards global understanding of plant defence against aphids -timing and dynamics of early *Arabidopsis* defence responses to cabbage aphid (*Brevicoryne brassicae*) attack, Plant Cell Environ. 31, 1097–1115.
- Labana K.S., Gupta M.L. (1993) Importance and Origin, in: Labana K.S., Banga S.S., Banga S.K. (Eds.), Breeding Oilseed Brassicas, Springer-Verlag, Berlin, Germany, pp. 1–7.
- Lamb R.J. (1989) Entomology of oilseed *Brassica* crops, Annu. Rev. Entomol. 34, 211–229.
- Landolt P. (1993) Effects of host plant leaf damage on cabbage looper moth attraction and oviposition, Entomol. Exp. Appl. 67, 79–85.
- Larkin R.P., Griffin T.S. (2007) Control of soilborne potato diseases using *Brassica* green manures, Crop Prot. 26, 1067–1077.
- Lauro N., Kuhlmann U., Mason P.G., Holliday N.J. (2005) Interaction of a solitary larval endoparasitoid *Microplitis mediator* with its host *Mamestra brassicae*: host acceptance and host suitability, J. Appl. Entomol. 129, 567–573.
- Law J.H., Regnier F.E. (1971) Pheromones, Annu. Rev. Biochem. 40, 533–548.
- LeCoz C., Ducombs G. (2006) Plants and plant products, in: Frosch P.J., Menne T., Lepottevin J.P. (Eds.), Contact Dermatitis, 4th ed., Springer Verlag, Berlin-Heidelberg, Germany, pp. 751–800.
- Leckband G., Frauen M., Friedt W. (2002) NAPUS 2000. Rapeseed (*Brassica napus*) breeding for improved human nutrition, Food Res. Int. 35, 273–278.
- Lehrman A. (2007) Oilseed rape transformed with a pea lectin gene: Target and non-target insects, plant competition, and farmer attitudes. Doctoral Thesis, Faculty of Natural Resources and Agricultural Sciences Department of Ecology, Swedish University of Agricultural Sciences, Uppsala, Sweden.
- Lenman M., Falk A., Rodin J., Höglund A.S., Ek B., Rask L. (1993) Differential expression of myrosinase gene families, Plant Physiol. 103, 703–711.
- Lewis W.J., Van Lenteren J.C., Phatak S.C., Tumlinson J.H. (1997) A total system approach to sustainable pest management, Proc. Natl Acad. Sci. (USA) 94, 12243–12248.
- Li Q., Eigenbrode S.D., Stringam G.R., Thiagarajah M.R. (2000) Feeding and growth of *Plutella xylostella* and *Spodoptera eridania* on *Brassica juncea* with varying glucosinolate concentrations and myrosinase activities, J. Chem. Ecol. 26, 2401–2419.
- Liu C.W., Lin C.C., Yiu J.C., Chen J.J.W., Tseng M.J. (2008) Expression of a *Bacillus thuringiensis* toxin (cry1Ab) gene in cabbage (*Brassica oleracea* L. var. *capitata* L.) chloroplasts confers high insecticidal efficacy against *Plutella xylostella*, Theor. Appl. Genet. 117, 75–88.

- Lou P., Zhao J., He H., Hanhart C., Del Carpio D.P., Verkerk R., Custers J., Koornneef M., Bonnema G. (2008) Quantitative trait loci for glucosinolate accumulation in *Brassica rapa* leaves, New Phytol. 179, 1017–1032.
- Ma W.C., Schoonhoven L.M. (1973) Tarsal contact chemosensory hairs of the large butterfly *Pieris brassicae* and their possible role in oviposition behaviour, Entomol. Exp. Appl. 16, 343–357.
- MacLeod A.J., Rossiter J.T. (1985) The occurrence and activity of epithiospecifier protein in some cruciferae seeds, Phytochemistry 24, 1895–1898.
- Mandal S.M.A., Mishra R.K., Patra A.K. (1994) Yield loss in rapeseed and mustard due to aphid infestation in respect of different cultivars and dates of sowing, Orissa J. Agric. Res. 7, 58–62.
- Marazzi C., Patrian B., Städler E. (2004a) Secondary metabolites of the leaf surface affected by sulphur fertilisation and perceived by the diamondback moth, Chemoecology 14, 81–86.
- Marazzi C., Patrian B., Städler E. (2004b) Secondary metabolites of the leaf surface affected by sulphur fertilisation and perceived by the cabbage root fly, Chemoecology 14, 87–94.
- Matthiessen J.N., Shackleton M.A. (2005) Biofumigation: environmental impacts on the biological activity of diverse pure and plant-derived isothiocyanates, Pest Manag. Sci. 61, 1043–1051.
- Mattiacci L., Dicke M., Posthumus M.A. (1994) Induction of parasitoid attracting synomone in brussels sprouts plants by feeding of *Pieris brassicae* larvae: Role of mechanical damage and herbivore elicitor, J. Chem. Ecol. 20, 2229–2247.
- Mattiacci L., Rudelli S., Rocca B.A., Genini S., Dorn S. (2001) Systemically-induced response of cabbage plants against a specialist herbivore, *Pieris brassicae*, Chemoecology 11, 167–173.
- MBTOC (1997) Report of the Methyl Bromide Technical Options Committee, Nairobi, Kenya, United Nations Environmental Programme UNEP, 221 p.
- Mello M.O., Silva-Filho M.C. (2002) Plant-insect interactions: an evolutionary arms race between two distinct defense mechanisms, Braz. J. Plant Physiol. 14, 71–81.
- Mewis I., Tokuhisa J.G., Schultz J.C., Appel H.M., Ulrichs C., Gershenzon J. (2006) Gene expression and glucosinolate accumulation in *Arabidopsis thaliana* in response to generalist and specialist herbivores of different feeding guilds and the role of defense signaling pathways, Phytochemistry 67, 2450–2462.
- Mithen R. (1992) Leaf glucosinolate profiles and their relationship to pest and disease resistance in oilseed rape, Euphytica 63, 71–83.
- Monfort W.S., Csinos A.S., Desaeger J., Seebold K., Webster T.M., Diaz-Perez J.C. (2007) Evaluating *Brassica* species as an alternative control measure of root-knot nematode (*M. incognita*) in Georgia vegetable plasticulture, Crop Prot. 26, 1359–1368.
- Montesano M., Brader G., Palva E.T. (2003) Pathogen derived elicitors: searching for receptors in plants, Mol. Plant Pathol. 4, 73–79.
- Mossler M.A. (2005) Florida Crop/Pest Management Profile: Specialty Brassicas (Arrugula, Bok Choy, Chinese Broccoli, Chinese Mustard, Napa) PI-70, Pesticide Information Office, University of Florida, USA.

- Mudd A., Ferguson A.W., Blight M.M., Williams I.H., Scubla P., Solinas M., Clark S.J. (1997) Extraction, isolation, and composition of oviposition-deterring secretion of cabbage seed weevil *Ceutorhynchus assimilis*, J. Chem. Ecol. 23, 2227–2240.
- Mulligan E.A., Ferry N., Jouanin L., Walters K.F., Port G.R., Gatehouse A.M. (2006) Comparing the impact of conventional pesticide and use of a transgenic pest-resistant crop on the beneficial carabid beetle *Pterostichus melanarius*, Pest Manag. Sci. 62, 999–1012.
- Mumm R., Posthumus M.A., Dicke M. (2008) Significance of terpenoids in induced indirect plant defence against herbivorous arthropods, Plant Cell Environ. 31, 575–585.
- Murchie A.K. Hume K.D. (2003) Evidence for monogeny in the brassica pod midge *Dasineura brassicae*, Entomol. Exp. Appl. 107, 237–241.
- Murchie A.K., Smart L.E., Williams I.H. (1997) Responses of *Dasineura brassicae* and its parasitoids *Platygaster subuliformis* and *Omphale clypealis* to field traps baited with organic isothiocyanates, J. Chem. Ecol. 23, 917–926.
- Müller C., Arand K. (2007) Trade-offs in oviposition choice? Food-dependent performance and defence against predators of a herbivorous sawfly, Entomol. Exp. Appl. 124, 153–159.
- Müller C., Brakefield P.M. (2003) Analysis of a chemical defense in sawfly larvae: easy bleeding targets predatory wasps in late summer, J. Chem. Ecol. 29, 2683–2694.
- Müller C., Sieling N. (2006) Effects of glucosinolate and myrosinase levels in *Brassica juncea* on a glucosinolatesequestering herbivore – and vice versa, Chemoecology 16, 191–201.
- Neveu N., Grandgirard J., Nenon J.P., Cortesero A.M. (2002) Systemic release of herbivore-induced plant volatiles by turnips infested by concealed root-feeding larvae *Delia radicum* L., J. Chem. Ecol. 28, 1717–1732.
- Nicolas S.D., Leflon M., Monod H., Eber F., Coriton O., Huteau V., Chèvre A.M., Jenczewski E. (2009) Genetic regulation of meiotic cross-overs between related genomes in *Brassica napus* haploids and hybrids, Plant Cell 21, 373–385.
- Noble R.R.P., Harvey S.G., Sams C.E. (2002) Toxicity of Indian mustard and allyl isothiocyanate to masked chafer beetle larvae, Plant Health Progress, DOI: 10.1094/PHP-2002-0610-01-RS.
- Noldus L.P.J.J., Van Lenteren J.C. (1985a) Kairomones for egg parasite *Trichogramma evanescens*, Westwood I. Effect of volatiles released by two of its hosts, *Pieris brassicae* L. and *Mamestra brassicae* L., J. Chem. Ecol. 11, 781–791.
- Noldus L.P.J.J., Van Lenteren J.C. (1985b) Kairomones for egg parasite *Trichogramma evanescens*, Westwood II. Effect of contact chemicals produced by two of its hosts, *Pieris brassicae* L. and *Pieris rapae* L., J. Chem. Ecol. 11, 793–798.
- Nottingham S.F., Coaker T.H. (1985) The olfactory response of cabbage root fly *Delia radicum* to the host plant volatile allylisothiocyanate, Entomol. Exp. Appl. 39, 307–316.
- Nürnberger T., Brunner F. (2002) Innate immunity in plants and animals: emerging parallels between the recognition of general elicitors and pathogen-associated molecular patterns, Curr. Opin. Plant Biol. 5, 318–324.
- Ode P.J. (2006) Plant chemistry and natural enemy fitness: Effects on herbivore and natural enemy interactions, Ann. Rev. Entomol. 51, 163–185.

- Ômura H., Honda K., Hayashi N. (1999) Chemical and chromatic bases for preferential visiting by the cabbage butterfly *Pieris rapae* to rape flowers, J. Chem. Ecol. 25, 1895–1906.
- Paré P.W., Tumlinson J.H. (1999) Plant volatiles as a defense against insect herbivores, Plant Physiol. 121, 325–332.
- Pedras M.S.C., Okanga F.I., Zaharia I.L., Khan A.Q. (2000) Phytoalexins from crucifers: synthesis, biosynthesis, and biotransformation, Phytochemistry 53, 161–176.
- Pedras M.S.C., Zheng Q.A., Gadagi R.S. (2007a) The first naturally occurring aromatic isothiocyanates, rapalexins A and B, are cruciferous phytoalexins, Chem. Commun. 4, 368–370.
- Pedras M.S.C., Gadagi R.S., Jha M., Sarma-Mamillapalle V.K. (2007b) Detoxification of the phytoalexin brassinin by isolates of *Leptosphaeria maculans* pathogenic on brown mustard involves an inducible hydrolase, Phytochemistry 68, 1572–1578.
- Pickett J.A., Wadhams L.J., Woodcock C.M. (1997) Developing sustainable pest control from chemical ecology, Agr. Ecosyst. Environ. 64, 149–156.
- Pickett J.A., Rasmussen H.B., Woodcock C.M., Matthes M., Napier J.A. (2003) Plant stress signalling: understanding and exploiting plant-plant interactions, Biochem. Soc. T. 31, 123–127.
- Picoaga A., Cartea M.E., Soengas P., Monetti L., Ordás A. (2003) Resistance of kale populations to lepidopterous pests in northwestern Spain, J. Econ. Entomol. 96, 143–147.
- Pivnick K.A., Lamb R.J., Reed D. (1992) Response of flea beetles, *Phyllotreta* spp., to mustard oils and nitriles in field trapping experiments, J. Chem. Ecol. 18, 863–873.
- Pivnick K.A., Jarvis B.J., Slater G.P. (1994) Identification of olfactory cues used in host-plant finding by diamondback moth, *Plutella xylostella* (Lepidoptera: Plutellidae), J. Chem. Ecol. 20, 1407–1427.
- Ploeg A. (2008) Biofumigation to manage plant-parasitic nematodes, in: Ciancio A., Mukerji K.G. (Eds.), Integrated Management of Plant Pests and Diseases, Vol. 2 - Integrated Management and Biocontrol of Vegetable and Grain Crops Nematodes, Springer, The Netherlands, pp. 239–248.
- Pontoppidan B., Hopkins R., Rask L. (2003) Infestation by cabbage aphid (*Brevicoryne brassicae*) on oilseed rape (*Brassica napus*) causes a long lasting induction of the myrosinase system, Entomol. Exp. Appl. 109, 55–62.
- Pope T.W., Kissen R., Grant M., Pickett J.A., Rossiter J.T., Powell G. (2008) Comparative innate responses of the aphid parasitoid *Diaeretiella rapae* to alkenyl glucosinolate derived isothiocyanates, nitriles and epithionitriles, J. Chem. Ecol. 34, 1302–1310.
- Popova T. (1993) A study of antibiotic effects on cabbage cultivars on the cabbage moth *Mamestra brassicae* L. (Lepidoptera: Noctuidae), Entomol. Rev. 72, 125–132.
- Potter M.J., Davies K., Rathjen A.J. (1998) Suppressive impact of glucosinolates in *Brassica* vegetative tissues on root lesion nematode *Pratylenchus neglectus*, J. Chem. Ecol. 24, 67–80.
- Pratt C., Pope T., Powell G., Rossiter J. (2008) Accumulation of glucosinolates by the cabbage aphid *Brevicoryne brassicae* as a defense against two coccinellid species, J. Chem. Ecol. 34, 323–329.
- Pyke B., Rice M., Sabine B., Zalucki M.P. (1987) The push-pull strategy - behavioural control of *Heliothis*, Aust. Cotton Grow, May–July, 7–9.

- Quaglia F., Rossi E., Petacchi R., Taylor C.E. (1993) Observations on an infestation by green peach aphids (Homoptera: Aphididae) on greenhouse tomatoes in Italy, J. Econ. Entomol. 86, 1019–1025.
- Rana J. (2005) Performance of *Lipaphis erysimi* (Homoptera: Aphididae) on different *Brassica* species in a tropical environment, J. Pestic. Sci. 78, 155–160.
- Rangkadilok N., Nicolas M.E., Bennett R.N., Premier R.R., Eagling D.R., Taylor P.W.J. (2002) Developmental changes of sinigrin and glucoraphanin in three *Brassica* species (*Brassica nigra*, *Brassica juncea* and *Brassica oleracea* var. *italica*), Sci. Hortic. 96, 11–26.
- Rask L., Andreasson E., Ekbom B., Eriksson S., Pontoppidan B., Meijer J. (2000) Myrosinase: gene family evolution and herbivore defence in Brassicaceae, Plant Mol. Biol. 42, 93–113.
- Rausch T., Wachter A. (2005) Sulfur metabolism: a versatile platform for launching defence operations, Trends Plant Sci. 10, 503–509.
- Raymer P.L. (2002) Canola: An emerging oilseed crop, in: Janwick J., Whipkey A. (Eds.), Trends in new crops and new uses, ASHS Press, Alexandria, VA, USA.
- Reddy G.V.P., Holopainen J.K., Guerrero A. (2002) Olfactory responses of *Plutella xylostella* natural enemies to host pheromone, larval fraa, and green leaf cabbage volatiles, J. Chem. Ecol. 28, 131–143.
- Reddy G.V.P., Tossavainen P., Nerg A.M., Holopainen J.K. (2004) Elevated atmospheric CO<sub>2</sub> affects the chemical quality of *Brassica* plants and the growth rate of the specialist *Plutella xylostella*, but not the generalist *Spodoptera littoralis*, J. Agr. Food Chem. 52, 4185–4191.
- Renwick J., Haribal M., Gouinguené S., Städler E. (2006) Isothiocyanates stimulating oviposition by the diamondback moth, *Plutella xylostella*, J. Chem. Ecol. 32, 755–766.
- Renwick J.A.A. (2001) Variable diets and changing taste in plant-insect relationships, J. Chem. Ecol. 27, 1063–1076.
- Renwick J.A.A., Lopez K. (1999) Experience-based food consumption by larvae of *Pieris rapae*: addiction to glucosinolates, Entomol. Exp. Appl. 91, 51–58.
- Renwick J.A.A., Radke C.D. (1980) An oviposition deterrent associated with frass from feeding larvae of the cabbage looper, *Trichoplusia ni* (Lepidoptera: Noctuidae), Environ. Entomol. 9, 318–320.
- Renwick J.A.A., Radke C.D. (1981) Host plant constituents as oviposition deterrents for the cabbage looper, Entomol. Exp. Appl. 30, 201–204.
- Renwick J.A.A., Radke C.D. (1985) Constituents of hostand non-host plants deterring oviposition by the cabbage butterfly, *Pieris rapae*, Entomol. Exp. Appl. 39, 21–26.
- Renwick J.A.A., Radke C.D., Sachdev-Gupta K., Städler E. (1992) Leaf surface chemicals stimulating oviposition by *Pieris rapae* (Lepidoptera: Pieridae) on cabbage, Chemoecology 3, 33–38.
- Riaño-Pachón D.M., Nagel A., Neigenfind J., Wagner R., Basekow R., Weber E., Mueller-Roeber B., Diehl S., Kersten B. (2009) GabiPD: The GABI primary database – a plant integrative 'omics' database, Nucleic Acids Res. 37, D954–959.
- Rodman J.E., Chew F.S. (1980) Phytochemical correlates of herbivory in a community of native and naturalized cruciferae, Biochem. Syst. Ecol. 8, 43–50.

- Rodriguez V.M., Cartea M.E., Padilla G., Velasco P., Ordas A. (2005) The nabicol: A horticultural crop in northwestern Spain, Euphytica 142, 237–246.
- Roessingh P., Städler E. (1990) Foliar form, colour and surface characteristics influence oviposition behaviour in the cabbage root fly *Delia radicum*, Entomol. Exp. Appl. 57, 93–100.
- Roessingh P., Städler E., Fenwick G., Lewis J., Nielsen J., Hurter J., Ramp T. (1992) Oviposition and tarsal chemoreceptors of the cabbage root fly are stimulated by glucosinolates and host plant extracts, Entomol. Exp. Appl. 65, 267–282.
- Roessingh P., Städler E., Baur R., Hurter J., Ramp T. (1997) Tarsal chemoreceptors and oviposition behaviour of the cabbage root fly (*Delia radicum*) sensitive to fractions and new compounds of host-leaf surface extracts, Physiol. Entomol. 22, 140–148.
- Rohloff J., Bones A.M. (2005) Volatile profiling of Arabidopsis thaliana – Putative olfactory compounds in plant communication, Phytochemistry 66, 1941–1955.
- Rojas J.C., Wyatt T.D. (1999) The role of pre- and post-imaginal experience in the host-finding and oviposition behaviour of the cabbage moth, Physiol. Entomol. 24, 83–89.
- Rojas J.C., Tristram D., Wyatt T.D., Birch M.C. (2000) Flight and oviposition behavior toward different host plant species by the cabbage moth, *Mamestra brassicae* (L.) (Lepidoptera: Noctuidae), J. Insect Behav. 13, 247–254.
- Rojas J.C., Wyatt T.D., Birch M.C. (2001) Oviposition by *Mamestra brassicae* (L.) (Lep., Noctuidae) in relation to age, time of day and host plant, J. Appl. Entomol. 125, 161–163.
- Ross-Craig S. (1949) Drawings of British plants: Part III. Cruciferae, G. Bells and Sons Ltd, London, UK.
- Rossiter J.T., Jones A.M., Bones A.M. (2003) A novel myrosinase-glucosinolate defense system in cruciferous specialist aphids, Recent Adv. Phytochem. 37, 127–142.
- Rouxel T., Kollmann A., Boulidard L., Mithen R. (1991) Abiotic elicitation of indole phytoalexins and resistance to *Leptosphaeria maculans* within Brassiceae, Planta 184, 271–278.
- Ruther J., Thiemann K. (1997) Response of the pollen beetle *Meligethes aeneus* to volatiles emitted by intact plants and conspecifics, Entomol. Exp. Appl. 84, 183–188.
- Ryan M.F. (2002) Plant chemicals in pest control, in: Ryan M.F. (Ed.), Insect Chemoreception: Fundamental and Applied, Springer, The Netherlands, pp. 193–222.
- Salem S.A., Abou-Ela R.G., Matter M.M., El-Kholy M.Y. (2007) Entomocidal effect of *Brassica napus* extracts on two store pests, *Sitophilus oryzae* (L.) and *Rhizopertha dominica* (Fab.) (Coleoptera), J. Appl. Sci. Res. 3, 317–322.
- Sanders D. (2005) Growers Guidelines. American Vegetable Grower, October 2005 University of California SAREP Online Cover Crop Database: Mustards, http://www.sarep. ucdavis.edu.
- Sang J.P., Minchinton I.R., Johnstone P.K., Truscott R.J.W. (1984) Glucosinolate profiles in the seed, root and leaf tissue of cabbage, mustard, rapeseed, radish and swede, Can. J. Plant Sci. 64, 77–93.
- Sarfraz M., Dosdall L.M., Keddie B.A. (2006) Diamondback moth-host plant interactions: Implications for pest management, Crop Prot. 25, 625–639.
- Sarfraz M., Dosdall L.M., Keddie B.A. (2007) Resistance of some cultivated Brassicaceae to infestations by *Plutella*

xylostella (Lepidoptera: Plutellidae), J. Econ. Entomol. 100, 215–224.

- Sarwar M., Kirkegaard J.A., Wong P.T.W., Desmarchelier J.M. (1998) Biofumigation potential of brassicas, III. *In vitro* toxicity of isothiocyanates to soil-borne fungal pathogens, Plant Soil 201, 103–112.
- Sato Y., Ohsaki N. (2004) Response of the wasp (*Cotesia glomerata*) to larvae of the large white butterfly (*Pieris brassicae*), Ecol. Res. 19, 445–449.
- Schoonhoven L.M. (1990) Host-marking pheromones in Lepidoptera, with special reference to two *Pieris* spp., J. Chem. Ecol. 16, 3043–3052.
- Schroeder N.C., Dumbleton A.J. (2001) Thiamethoxam seed coating on rape seed for the control of cabbage aphid *Brevicoryne brassicae* (L.), N. Z. Plant Protect. 54, 240–243.
- Schuler T.H., Denholm I., Jouanin L., Clark S.J., Clark A.J., Poppy G.M. (2001) Population-scale laboratory studies of the effect of transgenic plants on nontarget insects, Mol. Ecol. 10, 1845–1853.
- Schuler T.H., Denholm I., Clark S.J., Stewart C.N., Poppy G.M. (2004) Effects of *Bt* plants on the development and survival of the parasitoid *Cotesia plutellae* (Hymenoptera: Braconidae) in susceptible and *Bt*-resistant larvae of the diamondback moth, *Plutella xylostella* (Lepidoptera: Plutellidae), J. Insect Physiol. 50, 435–443.
- Sekhon B.S. (1999) Population dynamics of *Lipaphis erysimi* and *Myzus persicae* on different species of *Brassica*, 10th International Rapeseed Congress, Canberra, Australia.
- Sekhon B.S., Åhman I. (1993) Insect resistance with special reference to mustard aphid, in: Labana K.S., Banga S.S., Banga S.K. (Eds.), Breeding Oilseed Brassicas, Springer-Verlag, Berlin, Germany, pp. 206–217.
- Shelton A.M., Badenes-Perez F.R. (2006) Concepts and applications of trap cropping in pest management, Annu. Rev. Entomol. 51, 285–308.
- Shelton A.M., Hatch S.L., Zhao J.Z., Chen M., Earle E.D., Cao J. (2008) Suppression of diamondback moth using *Bt*-transgenic plants as a trap crop, Crop Prot. 27, 403–409.
- Shiojiri K., Takabayashi J., Yano S., Takafuji A. (2000) Herbivore-species-specific interactions between crucifer plants and parasitic wasps (Hymenoptera: Braconidae) that are mediated by infochemicals present in areas damaged by herbivores, Appl. Entomol. Zool. 35, 519–524.
- Shiojiri K., Takabayashi J., Yano S., Takafuji A. (2002) Oviposition preferences of herbivores are affected by tritrophic interaction webs, Ecol. Lett. 5, 186–192.
- Sibanda T., Dobson H.M., Cooper J.F., Manyangarirwa W., Chiimba W. (2000) Pest management challenges for smallholder vegetable farmers in Zimbabwe, Crop Prot. 19, 807–815.
- Silverstein K.A.T., Graham M.A., Paape T.D., VandenBosch K.A. (2005) Genome organization of more than 300 defensin-like genes in *Arabidopsis*, Plant Physiol. 138, 600–610.
- Silverstein K.A.T., Moskal Jr. W.W., Wu H.C., Underwood B.A., Graham M.A., Town C.D., VandenBosch K.A. (2007) Small cysteine-rich peptides resembling antimicrobial peptides have been under-predicted in plants, Plant J. 51, 262–280.
- Simmonds M., Blaney W., Mithen R., Birch A., Lewis J. (1994) Behavioural and chemosensory responses of the turnip root fly (*Delia floralis*) to glucosinolates, Entomol. Exp. Appl. 71, 41–57.

- Smallegange R., Van Loon J., Blatt S., Harvey J., Agerbirk N., Dicke M. (2007) Flower vs. leaf feeding by *Pieris brassicae*: Glucosinolate-rich flower tissues are preferred and sustain higher growth rate, J. Chem. Ecol. 33, 1831–1844.
- Smart L.E., Blight M.M. (1997) Field discrimination of oilseed Rape, *Brassica napus* volatiles by cabbage seed weevil, *Ceutorhynchus assimilis*, J. Chem. Ecol. 23, 2555–2567.
- Smart L.E., Blight M.M. (2000) Response of the pollen beetle, *Meligethes aeneus*, to traps baited with volatiles from oilseed rape, *Brassica napus*, J. Chem. Ecol. 26, 1051–1064.
- Smart L.E., Blight M.M., Hick A.J. (1997) Effect of visual cues and a mixture of isothiocyanates on trap capture of cabbage seed weevil, *Ceutorhynchus assimilis*, J. Chem. Ecol. 23, 889–902.
- Smith B.J., Kirkegaard J.A. (2002) In vitro inhibition of soil microorganisms by 2-phenylethyl isothiocyanate, Plant Pathol. 51, 585–593.
- Smith B.J., Kirkegaard J.A., Howe G.N. (2004) Impacts of *Brassica* break-crops on soil biology and yield of following wheat crops, Aust. J. Agr. Res. 55, 1–11.
- Smith M.L. (2002) Research in Scottland The Scottish Agricultural College (SAC): An overview, Pesticide Outlook, 59–60.
- Smith M.C. (2005) Plant Resistance to Arthropods Molecular and conventional approaches, Chap. 2: Antixenosis
  Adverse effects of resistance on arthropod behavior, Springer, Dordrecht, The Netherlands, pp. 19–63.
- Snapp S., Cichy K., O'Neil K. (2006) Mustards A Brassica cover crop for Michigan, Extension Bulletin E-2956, Michigan State University Extension, USA.
- Snowdon R.J. (2007) Cytogenetics and genome analysis in *Brassica* crops, Chromosome Res. 15, 85–95.
- Snowdon R.J., Friedt W. (2004) Molecular markers in *Brassica* oilseed breeding: current status and future possibilities, Plant Breeding 123, 1–8.
- Snowdon R.J., Luhs W., Friedt W. (2007) Oilseed rape, in: Kole C. (Ed.), Genome Mapping and Molecular Breeding in Plants, Vol. 2 Oilseeds, Springer-Verlag Berlin Heidelberg, Germany, pp. 55–114.
- Souckova H. (2006) Rape methyl-ester as a renewable energy resource in transport, Agr. Econ.-Czech. 52, 244–249.
- Steinberg S., Dicke M., Vet L.E.M. (1993) Relative importance of infochemicals from first and second trophic level in long-range host location by the larval parasitoid *Cotesia* glomerata, J. Chem. Ecol. 19, 47–59.
- Suwabe K., Tsukazaki H., Iketani H. Hatakeyama K., Kondo M., Fujimura M., Nunome T., Fukuoka H., Hirai M., Matsumoto S. (2006) Simple sequence repeat-based comparative genomics between *Brassica rapa* and *Arabidopsis thaliana*: The genetic origin of clubroot resistance, Genetics 173, 309–319.
- Svanem P., Bones A.M., Rossiter J.T. (1997) Metabolism of  $[\alpha^{-14}C]$ -desulphophenethylglucosinolate in *Nasturtium officinale*, Phytochemistry 44, 1251–1255.
- Tabashnik B.E. (1987) Plant secondary compounds as oviposition deterrents for cabbage butterfly, *Pieris rapae* (Lepidoptera: Pieridae), J. Chem. Ecol. 13, 309–316.
- Takabayashi J., Dicke M. (1996) Plant-carnivore mutualism through herbivore-induced carnivore attractants, Trends Plant Sci. 1, 109–113.

- Talekar N.S., Shelton A.M. (1993) Biology, ecology, and management of the diamondback moth, Annu. Rev. Entomol. 38, 275–301.
- Tanaka S., Nishida T., Ohsaki N. (2007) Sequential rapid adaptation of indigenous parasitoid wasps to the invasive butterfly *Pieris brassicae*, Evolution 61, 1791–1802.
- Thangstad O.P., Iversen T.H., Slupphaug G., Bones A. (1990) Immunocytochemical localization of myrosinase in *Brassica napus* L., Planta 180, 245–248.
- Thangstad O.P., Evjen K., Bones A.M. (1991) Immunogold-EM localization of myrosinase, Protoplasma 161, 85–93.
- Thangstad O.P., Winge P., Husebye H., Bones A.M. (1993) The thioglucoside glucohydrolase (myrosinase) gene family in Brassicacea, Plant Mol. Biol. 23, 511–524.
- Thangstad O.P., Bones A.M., Holtan S., Moen L., Rossiter J.T. (2001) Microautoradiographic localisation of a glucosinolate precursor to specific cells in *Brassica napus* L. embryos indicates a separate transport pathway into myrosin cells, Planta 213, 207–213.
- Thangstad O.P., Gilde B., Chadchawan S., Seem M., Husebye H., Bradley D., Bones A.M. (2004) Cell specific, cross-species expression of myrosinases in *Brassica napus*, *Arabidopsis thaliana* and *Nicotiana tabacum*, Plant Mol. Biol. 54, 597–611.
- Thomma B., Cammue B., Thevissen K. (2002) Plant defensins, Planta 216, 193–202.
- Traynier R.M.M. (1967a) Stimulation of oviposition by the cabbage root fly *Erioischia brassicae*, Entomol. Exp. Appl. 10, 401–412.
- Traynier R.M.M. (1967b) Effect of host plant odour on the behaviour of the adult cabbage root fly, *Erioischia brassicae*, Entomol. Exp. Appl. 10, 321–328.
- Traynier R.M.M., Truscott R.J.W. (1991) Potent natural egglaying stimulant for cabbage butterfly *Pieris rapae*, J. Chem. Ecol. 17, 1371–1380.
- Tripathi M.K., Mishra A.S. (2007) Glucosinolates in animal nutrition: A review, Anim. Feed Sci. Tech. 132, 1–27.
- Tscharntke T., Hawkins B.A. (2002) Multitrophic level interactions, Cambridge University Press, Cambridge, UK.
- Tuttle A.F., Ferro D.N., Idoine K. (1988) Role of visual and olfactory stimuli in host finding of adult cabbage root flies, *Delia radicum*, Entomol. Exp. Appl. 47, 37–44.
- U.N. (1935) Genomic analysis in *Brassica* with special reference to the experimental formation of *B. napus* and peculiar mode of fertilization, Japan J. Bot. 7, 389–452.
- Ulland S., Ian E., Mozuraitis R., Borg-Karlson A.K., Meadow R., Mustaparta H. (2008) Methyl salicylate, identified as primary odorant of a specific receptor neuron type, inhibits oviposition by the moth *Mamestra brassicae* L. (Lepidoptera, Noctudae), Chem. Senses 33, 35–46.
- Ulmer B.J. (2002) Crucifer host plant suitability for bertha armyworm (*Mamestra configrata*) and diamondback moth (*Plutella xylostella*). Doctoral thesis, College of Graduate Studies and Research, Department of Biology, University of Saskatchewan, Saskatoon, Saskatchewan, Canada.
- Ulmer B.J., Gillott C., Woods D., Erlandson M. (2002) Diamondback moth, *Plutella xylostella* (L.), feeding and oviposition preferences on glossy and waxy *Brassica rapa* (L.) lines, Crop Prot. 21, 327–331.
- Ulmer B.J., Gillott C., Erlandson M. (2003) Conspecific eggs and bertha armyworm, *Mamestra configrata* (Lepidoptera:

Noctuidae), Oviposition site selection, Environ. Entomol. 32, 529–534.

- Ulmer B.J., Dosdall L.M. (2006) Glucosinolate profile and oviposition behavior in relation to the susceptibilities of Brassicaceae to the cabbage seedpod weevil, Entomol. Exp. Appl. 121, 203–213.
- Valantin-Morison M., Meynard J.M., Dore T. (2007) Effects of crop management and surrounding field environment on insect incidence in organic winter oilseed rape (*Brassica napus* L.), Crop Prot. 26, 1108–1120.
- Van Dam N.M., Raaijmakers C.E., Van der Putten W.H. (2005) Root herbivory reduces growth and survival of the shoot feeding specialist *Pieris rapae* on *Brassica nigra*, Entomol. Exp. Appl. 115, 161–170.
- Van Dam N., Raaijmakers C. (2006) Local and systemic induced responses to cabbage root fly larvae (*Delia radicum*) in *Brassica nigra* and *B. oleracea*, Chemoecology 16, 17–24.
- Van Loon J.J.A., Blaakmeer A., Griepink F.C., Van Beek T.A., Schoonhoven L.M., De Groot A. (1992) Leaf surface compound from *Brassica oleracea* (Cruciferae) induces oviposition by *Pieris brassicae* (Lepidoptera: Pieridae), Chemoecology 3, 39–44.
- Van Loon J.J.A., Wang C.Z., Nielsen J.K., Gols R., Qiu Y.T. (2002) Flavonoids from cabbage are feeding stimulants for diamondback moth larvae additional to glucosinolates: Chemoreception and behaviour, Entomol. Exp. Appl. 104, 27–34.
- Verhoeven D.T.H., Verhagen H., Goldbohm R.A., Van den Brandt P.A., Van Poppel G. (1997) A review of mechanisms underlying anticarcinogenicity by *Brassica* vegetables, Chem.-Biol. Interact. 103, 79–129.
- Verkerk R.H.J., Wright D.J. (1996) Common cabbage resistance mechanisms against the diamondback moth: still an open book, Ann. Appl. Biol. 128, 571–577.
- Verschaffelt E. (1910) De oorzaak der voedselkeus bij eenige plantenetende insecten, Versl. gewone Vergad. Akad. Amst. 19, 594–600.
- Vet L.E.M., Dicke M. (1992) Ecology of infochemical use by natural enemies in a tritrophic context, Ann. Rev. Entomol. 37, 141–172.
- Vig K. (2002) Data on the biology of the crucifer flea beetle, *Phyllotreta cruciferae* (Goeze, 1777) (Coleoptera, Chrysomelidae, Alticinae), Mededelingen (Rijksuniversiteit te Gent. Fakulteit van de Landbouwkundige en Toegepaste Biologische Wetenschappen) 67, 537–546.
- Visvalingam S., Hønsi T.G., Bones A.M. (1998) Sulphate and micronutrients can modulate the expression levels of myrosinases in *Sinapis alba* L. plants, Physiol. Plant. 104, 30–37.
- Wang J., Chen Z., Du J., Sun Y., Liang A. (2005) Novel insect resistance in *Brassica napus* developed by transformation of chitinase and scorpion toxin genes, Plant Cell Rep. 24, 549–555.
- Warton B., Matthiessen J.N., Shackleton M.A. (2003) Crossenhancement: enhanced biodegradation of isothiocyanates in soils previously treated with metham sodium, Soil Biol. Biochem. 35, 1123–1127.
- Williams I.H. (2006) Integrating parasitoids into management of pollen beetle on oilseed rape, Agron. Res. 4 (Special Issue), 465–470.

- Williams J.S., Cooper R.M. (2004) The oldest fungicide and newest phytoalexin - a reappraisal of the fungitoxicity of elemental sulphur, Plant Pathol. 53, 263–279.
- Wittstock U., Halkier B.A. (2002) Glucosinolate research in the Arabidopsis era, Trends Plant Sci. 7, 263–270.
- Wittstock U., Agerbirk N., Stauber E.J., Olsen C.E., Hippler M., Mitchell-Olds T., Gershenzon J., Vogel H. (2004) Successful herbivore attack due to metabolic diversion of a plant chemical defense, Proc. Natl Acad. Sci. (USA) 101, 4859–4864.
- Wolfson J. (1980) Oviposition response of *Pieris rapae* to environmentally induced variation in *Brassica nigra*, Entomol. Exp. Appl. 27, 223–232.
- Wright D.J., Verkerk R.H.J. (1995) Integration of chemical and biological control systems for arthropods: Evaluation in a multitrophic context, Pestic. Sci. 44, 207–218.

- Xue J., Lenman M., Falk A., Rask L. (1992) The glucosinolatedegrading enzyme myrosinase in Brassicaceae is encoded by a gene family, Plant Mol. Biol. 18, 387–398.
- Zabala M.De.T., Grant M., Bones A.M., Bennett R., Lim Y.S., Kissen R., Rossiter J.T. (2005) Characterisation of recombinant epithiospecifier protein and its over-expression in *Arabidopsis thaliana*, Phytochemistry 66, 859–867.
- Zhang G., Zhou W. (2006) Genetic analyses of agronomic and seed quality traits of synthetic oilseed *Brassica napus* produced from interspecific hybridization of *B. campestris* and *B. oleracea*, J. Genet. 85, 45–51.
- Zhou W.J., Zhang L.W. (2001) Oilseed rape, in: Wu J.L., Zhang G.P. (Eds.), Seed Production and Quality Control in Crops, Zhejiang University Press, Hangzhou, China, pp. 78–95.
- Zukalová H., Vašák J. (2002) The role and effects of glucosinolates of *Brassica* species - A review, Rost. Výroba 48, 175–180.

# Ionising Radiation and Area-Wide Management of Insect Pests to Promote Sustainable Agriculture

Marc J.B. Vreysen and Alan S. Robinson

## Contents

Introduction: Insecticide Use and Food Security				
Area Mana	Wide Integrated Pest agement (AW-IPM)	673		
Steril	e Insects	674		
3.1	Rearing Insects	676		
3.2	Producing Sterile Males	676		
3.3	Inducing Sterility	678		
3.4	Sterile Insect Quality Management	680		
3.5	Monitoring Insects	680		
3.6	Using Sterile Insects	681		
4 Ionising Radiation and Natural Enemies				
4.1	Ionising Radiation for Improved Rearing	682		
4.2	Ionising Radiation to Facilitate Handling,			
	Shipment, Trade and Release	683		
4.3	Integrating the Sterile Insect Technique (SIT)			
	or F <sub>1</sub> Sterility and Natural Enemies	683		
Prote	cting the Environment	683		
6 Enhancing Sustainable Agriculture				
6.1	New World Screwworm in the Americas	685		
6.2	Chile's National Fruit Fly Programme	685		
6.3	Melon Fly in Okinawa	685		
6.4	Cassava Mealy Bug in Africa	686		
6.5	The Private Sector Enters the Fruit Fly			
	Industry in Israel	686		
6.6	Codling Moth in Canada	686		
7 The Future				
Conc	lusions	687		
References				
	Intro Area- Mana Steril 3.1 3.2 3.3 3.4 3.5 3.6 Ionisi 4.1 4.2 4.3 Prote Enha 6.1 6.2 6.3 6.4 6.5 6.6 The I Conc	Introduction: Insecticide Use and Food Security         Area-Wide Integrated Pest         Management (AW-IPM)         Sterile Insects         3.1       Rearing Insects         3.2       Producing Sterile Males         3.3       Inducing Sterile Males         3.3       Inducing Sterile Males         3.3       Inducing Sterile Males         3.3       Inducing Sterile Males         3.3       Inducing Sterile Males         3.4       Sterile Insect Quality Management         3.5       Monitoring Insects         3.6       Using Sterile Insects         4.1       Ionising Radiation and Natural Enemies         4.2       Ionising Radiation to Facilitate Handling, Shipment, Trade and Release         4.3       Integrating the Sterile Insect Technique (SIT) or F1 Sterility and Natural Enemies         Protecting the Environment         Enhancing Sustainable Agriculture         6.1       New World Screwworm in the Americas         6.2       Chile's National Fruit Fly Programme         6.3       Melon Fly in Okinawa         6.4       Cassava Mealy Bug in Africa         6.5       The Private Sector Enters the Fruit Fly Industry in Israel         6.6       Codling Moth in Canada         The Fu		

Abstract Despite the liberal use of broad-spectrum insecticides to keep many insect pests of agricultural and veterinary importance at bay, food losses, both preand post-harvest, due to these insect pests contribute significantly to the high prevalence of undernourishment in the world. New, innovative pest control tactics and strategies are therefore needed that are both effective and not detrimental to the environment. As part of the arsenal of environmentally-friendly control tactics, the sterile insect technique (SIT) has proven to be a very effective tool against selected insect pests when used as part of an area-wide integrated pest management (AW-IPM) approach. Likewise, the use of natural enemies for augmentative or inundative biological control is now a major component of pest control in many parts of the world. Both control tactics are complementary and even synergistic under certain circumstances, but their combined use has so far not been applied on an operational scale. Ionising radiation can be readily employed to effectively and safely induce sexual sterility in insects. Although the sterile insect technique has often been associated with an eradication strategy, major advances in rearing efficiency, and improved handling and release methods, have made the use of sterile insects economically feasible for insect pest suppression, prevention or containment. Recently, more emphasis has been placed on the quality of the sterile insect once released in the field rather than mainly assessing quality in the rearing facility. This combined with other innovations such as the development of genetic sexing strains, better understanding the impact of radiation on radio-resistant species such as Lepidoptera and the development of the F<sub>1</sub> sterility concept, advances in monitoring the induced sterility, etc. have significantly increased the

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_29,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Vreysen and Robinson, Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2010009

M.J.B. Vreysen (S)

Insect Pest Control Laboratory, FAO/IAEA Agriculture and Biotechnology Laboratories, Joint FAO/IAEA Programme, Vienna, Austria e-mail: M.Vreysen@iaea.org

efficiency of the sterile insect technique for several insect species. The action of sterile insects is inversely dependent on the density of the target population, and sterile insects have the intrinsic capacity to actively search for and mate with the last individuals of a pest population. These two characteristics make them ideal to deal with outbreaks of invasive insect pests. The use of sterile insects presents no threat to the environment, but aspects such as diet and waste disposal in large rearing facilities or bio-security in cases where the rearing facility is located in an area that is already free of the pest require the necessary attention. Ionising radiation can also be applied to greatly improve the efficiency of mass-rearing, handling and shipment of insect parasitoids and predators. Area-wide integrated pest management programmes that use sterile insects or natural enemies are complex and management-intensive, and require a management structure that is exclusively dedicated to the programme. Past and current examples have shown the enormous benefit-cost ratios that these programmes can generate and their importance for enhanced agriculture is increasing in significance.

**Keywords** Natural enemies • Sterile insect technique • Area-wide integrated pest management • Economic benefits

# 1 Introduction: Insecticide Use and Food Security

Although the prevalence of undernourishment (i.e. those people whose dietary energy consumption is continuously below a minimum dietary energy requirement for maintaining a healthy life and carrying out light physical activity with an acceptable minimum body-weight for attained height) has declined worldwide from 18% in 1990–1992 to 13% in 2003–2005, the total number of undernourished people in the world has remained constant, or has even slightly increased (841 million in 1990–1992 to 848 million in 2003– 2005) (FAO, 2008). Even in the developed world, the Food and Agriculture Organization of the United Nations (FAO) estimates that more than 15 million people remain undernourished, and the US Department of Agriculture (USDA) estimates that 36.2 million US citizens, including 12.4 million children, were food-insecure in 2007 (US Working Group on the Food Crisis, 2009). In 2003–2005, the food deficit of the undernourished population was 90–110 kcal/person/day in countries such as Israel, Luxembourg, Canada, Norway, etc., but between 350 and 440 kcal/person/day in Liberia, Eritrea, Burundi, Sierra Leone, Haiti and the Democratic Republic of Congo (FAO, 2008).

Insect pests that harm livestock and damage crops, both pre- and post-harvest, are often at the root of food insecurity, with more than 10 000 insect and mite pests described. To keep these pests at bay, farmers use three million metric tons of insecticides annually at a cost of USD 35 000 million (Pimentel, 2007). In Europe alone, more than 140 000 tonnes of pesticides are sprayed on food crops every year, which corresponds to 280 grams/EU citizen/year (Reuters, 2008). Rising concerns about pesticide residues in food commodities, as well as increased resistance of many pest insects to insecticides, contamination of the environment and outbreaks of secondary pests, have compelled the EU to call for a ban on aerial cropspraying of 22 toxic chemicals by 2016 (United Press International, 2008). Therefore, enhancing more efficient livestock and crop production systems requires the development and implementation of insect pest control methods and strategies that are more effective, more target-specific, and which have minimal impact on the environment. During the last decades, numerous environmentally-friendly surrogates for the use of broad-spectrum synthetic insecticides have been developed, but their wide application has not always been straightforward as some of these techniques, such as mating disruption for lepidopteran pests (Cardé and Minks, 1995) or the sterile insect technique (SIT) (Dyck et al., 2005), require application on an area-wide basis (see below) (Vreysen et al., 2007a; Koul et al., 2008). Especially in Europe, farmers are very individualistic in their approach to pest control, and area-wide practices have not gained the same level of importance as in other continents. In the last decades, it has become clear that the careful selection of control tactics and their appropriate application and combination in space and time results in most cases in more effective pest control and thus enhanced sustainable agriculture (Vreysen et al., 2007a).

# 2 Area-Wide Integrated Pest Management (AW-IPM)

Many years ago, the innovative Knipling (1959, 1972, 1979) used simple mathematical models to show the importance of applying pest control techniques on an area-wide basis, targeting the entire pest population within a defined area, as opposed to a field-by-field approach where portions of the pest population are left untouched. In one of his models, he compared two areas with equivalent populations undergoing a five-fold natural rate of increase per year. In one area, 99% of 90% of the pest population is destroyed each year (but no control is conducted on the remaining 10%) while in the other area the control is less intense, eliminating only 90% of the pest population but from the total area (100%) each year. Following five years of these treatments in the two areas the model indicates that there would be 100 times more pests in the first area than in the second (Fig. 1). Exercising less intensive control against the total pest population is therefore more effective than subjecting only part of the population to more intensive control efforts (Klassen, 2005). These calculations, however oversimplified, illustrate how critical it is to consider the total pest population (areawide population control or total population control) in operational planning of pest control activities and

how devastating the recruitment of individuals from relic foci can be for the control effort (Knipling, 1979; Hendrichs et al., 2007). This paper will focus on the use of biological control agents, namely sterile insects and natural enemies, in area-wide integrated pest management programmes and illustrate the use of ionising radiation in the application of these biological control agents. Sterile insects were not previously considered biological control agents under the International Plant Protection Convention (IPPC) as they were not selfreplicating organisms, and were thus excluded from regulatory approval within the relevant International Standards for Phytosanitary Measures (ISPM). This situation changed in 2005 with the adoption by the IPPC of ISPM 3 "Guidelines for the Export, Shipment, Import and Release of Biological Control Agents and Other Beneficial Organisms" which recognised sterile insects as biological control agents (FAO, 2005).

Although sterile insects and natural enemies interact with the pest population in different ways, natural enemies by increasing daily mortality and sterile insects by reducing fertility, both kinds of agents must be distributed in the field on an area-wide basis. As well as sharing a similarity in the way they are applied, the biological control agents also have other components in common. The use of both biological control agents and sterile insects relies on efficient mass-rearing



**Fig. 1** (Right) the effect of total population control (less intense control (90%) on the total (100%) population) on the size and growth of a hypothetical insect population as compared with

(left) a strategy where only part of the population is subjected to control (99% control on 90% of the population) (Modified after Knipling, 1979)

technology and on protocols to maintain strain quality. Both types of agents must be released in large numbers at regular intervals. Use of both kinds of agents requires effective monitoring of released and wild insects before, during and after releases. Finally, the use of both kinds of agents benefits from an ability to manipulate the sex ratio of the released insects. As well as these similarities, they also have the potential to act synergistically to reduce pest numbers when used in combination with each other (Knipling, 1966, 1972; Carpenter et al., 2005).

## 3 Sterile Insects

The use of sterile insects for pest control was conceived in 1937 and first applied in 1954, when there was little concern for environmental issues, sustainability and invasive species in relation to the control of agricultural pests (Knipling, 1955). In fact, the motivation to develop this technology came from the observation that a major pest of cattle in the USA, the New World screwworm Cochliomyia hominivorax (Coquerel), was present in the field at very low densities compared with other insect pests, which would make it vulnerable to introduction of sterility. Knipling's conceptual breakthrough was to realise that insect pests could be controlled by compromising their hereditary machinery and that the integrity of this machinery is critical for the survival of the insect. As is the case with many new concepts, "this approach to screwworm control was proposed to a number of scientists, but little enthusiasm for its possibilities was generated" (Knipling, 1959). Although the approach has gained much credibility in the intervening years (Krafsur, 1998) there are still many scientists and administrators who do not fully appreciate both the limitations and the potential of the technique (e.g. Vale and Torr, 2005). The concept was simple but revolutionary and, like most paradigm shifts, is obvious with hindsight.

The approach was a radical departure from the traditional methods of pest control, which usually relied on compromising the survival of insect pest populations through manipulation of the insect's environment using insecticides (Courschee, 1960), removal of hosts (Kovaleski, 2007), or other aspects critical for the survival of the insect (Ford et al., 1970). Also, in contrast to the traditional approaches, sterile insects have no immediate impact on the pest population as they do not directly kill the insects in the field. It was also the first time that a population of a given species was used to control or suppress a field population of the same species. The sterile insect technique is truly speciesspecific, in that it exerts its effect through conspecific matings of released sterile insects and their wild counterparts. Simply stated, the sterile insect technique is a form of birth control.

When sterile insects were first released on the island of Curaçao, they were used in the absence of other control methods as a stand-alone technique, in part to be able to measure their effectiveness without the complication of other confounding treatments (Baumhover et al., 1955). Subsequent experience has shown that this is rarely the case and to be effective, sterile insects must be integrated with other control tactics as part of area-wide integrated pest management programmes (see Fig. 2, inverse density dependence) (Vreysen et al., 2007a).

Each pest control tactic has its limitations, be they biological, operational, ethical, social, environmental, economic, etc., and in that respect, the use of sterile insects is not an exception. Sterile insects cannot be used for species where the stage to be released, usually the adult, is the one responsible for damage to the crop or livestock, i.e. insect vectors of diseases (e.g. horn flies - but see below), species where the adults are plant herbivores (e.g. locusts) and species where adults cause nuisance (e.g. house fly, cockroaches). A second limitation of the sterile insect technique is its substantial start-up and operational costs. There-



**Fig. 2** The effectiveness of various control tactics in relation to the density of the pest population (pour on = application of residual insecticides on livestock). SIT: sterile insect technique

fore, the application of the sterile insect technique is largely limited to use against very important key insect pests for which effective and affordable alternative controls are not available (Lance and McInnis, 2005). In some cases there is a value-added component, as controlling the key pest without the use of chemicals provides the opportunity for the secondary pest species to be regulated by their natural enemies. Economics will also play a key role related to the cost of rearing and releasing large numbers of sterile insects. Methods to artificially rear many species of insects have not yet been developed, including for many species of beetles and weevils, the human botfly Dermatobia hominis (Linnaeus Jr.) and other parasites of mammals, and some insects with an obligatory diapause such as the western cherry fruit fly Rhagoletis indifferens Curran (Lance and McInnis, 2005).

Sterile insects can be used in several different types of programmes (see boxes) but their use has been and in many instances still is - associated with the concept of eradication/elimination of pest populations from ecologically isolated regions (Hendrichs et al., 2005). This is perhaps logical and a direct result of the successful eradication of New World screwworm populations from the USA, Mexico and Central America (Meyer, 1994; Wyss, 2000; Vargas-Terán, 2005). Sterile insects have a unique biological advantage that matches them very well to the concept of eradication, i.e. their effectiveness increases as the pest population declines in numbers: their action is inversely dependent on the density of the target population (Dame, 1970). Once an isolated pest population begins to decline in numbers, following initial releases of a given number of sterile insects, continued releases at the same rate will result in increased ratios of sterile to wild males, which will not only accelerate that decline but will inevitably lead to elimination (Knipling, 1979). This is in contrast to many other pest control methods - such as insecticides and many biological control agents - where each treatment eliminates a constant proportion of the target population and which makes them much more effective at high pest population densities (Dame, 1970) than sterile insects. Consequently, the optimal efficiency in an area-wide integrated pest management programme is therefore obtained when control tactics are specifically chosen for their effectiveness at different pest population densities and used in sequence in order to maximise their effectiveness (Fig. 2).

Complete ecological isolation of a target population is not a requirement to achieve eradication providing the appropriate operational strategy is chosen, i.e. the rolling carpet principle (Hendrichs et al., 2005). In this strategy, the target area is divided into different treatment blocks, which are selected so as to minimise reinvasion pressure. Control is exercised along a unidirectional front and the four operational phases of areawide integrated pest management (pre-intervention, population reduction, sterile male releases and maintenance/verification of low-prevalence/pest-free areas) are carried out simultaneously and in a phased manner. Therefore, once the programme has started, it cannot be interrupted until a pest-free area (in an eradication strategy) or the maintenance phase to sustain an area of low pest prevalence (in a suppression strategy) has been reached (Hendrichs et al., 2005). The New World screwworm programme in the Americas is an example of the rolling carpet principle. The programme started in the southern part of the USA in the 1950s and the control front advanced in the next 40 years through Mexico and the various countries of Central America, reaching Panama in the late 1990s (Wyss, 2000). Following the declaration of eradication in Panama, a permanent buffer zone of 30 000 km<sup>2</sup> was established in the Darien gap of Panama to protect the screwwormfree areas. Forty million sterile flies are released per week in the buffer zone to prevent reinvasion of the fly from South America (APHIS/USDA, 2001).

The intrinsic capability of sterile insects to actively search for and mate with the last individuals of a pest population and their increased effectiveness at low pest population densities makes them an ideal tool to deal with outbreaks of invasive pests (Vreysen et al., in preparation). Sterile insects were a component of programmes to eliminate biological incursions of the New World screwworm in Libya (Lindquist et al., 1992), the invasive Australian painted apple moth Teia anartoides Walker in New Zealand (Suckling et al., 2007), melon fly Bactrocera cucurbitae (Coquillett) in Okinawa, Japan (Koyama et al., 2004), and cactus moth Cactoblastis cactorum (Berg) in Mexico (NAPPO, 2008). New pest incursions are characterised, initially at least, by low population densities, and adequate overflooding ratios of sterile to wild insects can often be easily obtained if adequate laboratory rearing protocols can be established. As new pest introductions usually have a patchy or fragmented distribution, the release of sterile insects is amongst the most effective ways of finding and eradicating these populations; much more effective than other control tactics (e.g. deployment of toxic baits) that depend on the judgement of entomologists or field technicians (Vreysen et al., in preparation).

Sustainability is a key concern for all pest control programmes. The state of eradication is, by definition, a sustainable end point as there are no individuals of the targeted pest species left in the ecologically isolated target area. The question is whether the sustainable end point can be sustained. In certain circumstances, this is not easy to achieve as areas can be re-infested either accidentally or deliberately and an extensive monitoring system is required to confirm that the area remains pest-free (Barclay et al., 2005). This is not only expensive, but maintaining support and funding over time can be difficult, leading to a reduction in vigilance and an increasing chance of reinfestation. However, programmes such as the elimination of the tsetse fly Glossina austeni Newstead from Unguja Island, Zanzibar (Vreysen et al., 2000) or the removal of the New World screwworm from Libya (Lindquist et al., 1992) have demonstrated that eradication programmes can be sustainable. Permanent vigilance is required to prevent re-introduction and sometimes, stringent measures such as the continuous release of sterile flies on the islands close to Taiwan in the Okinawa archipelago or in the Darien gap in Panama are needed to protect the cleared areas (Koyama et al., 2004; Hendrichs et al., 2005).

In the last decade, sterile insects have increasingly been considered as part of area-wide integrated pest management suppression programmes against certain strategic pest insects. This is due to the fact that the rearing and release of sterile insects such as the Mediterranean fruit fly Ceratitis capitata (Wiedemann) has become much more cost-effective, and the use of insecticides has become much more restricted and more complex due to intermingling of commercial production areas with human settlements, increased demand for organic food, and difficulties in establishing effective quarantine measures to maintain an area pest-free (Hendrichs et al., 2005). A suppression strategy requires significantly lower investment in monitoring and quarantine measures need not be so stringent as compared to those required for eradication. These trends have resulted in more frequent use of sterile insects for suppressing populations of the Mediterranean fruit fly as an alternative to the use of insecticides (Barnes et al., 2004). This strategy has

mainly gained acceptance for use against crop pests, as a certain level of crop damage can be tolerated; something that is much more difficult with pests of veterinary or human health importance (Hendrichs et al., 2005). The various strategies are described in more detail in Section 3.6 and in Box 1.

## 3.1 Rearing Insects

The ability to rear the target insect using an artificial diet was one of the essential factors that enabled the use of sterile insects for New World screwworm control (Brown, 1984; Marroquin, 1985). Rearing insects for sterile release requires firstly the development of an artificial diet followed by upscaling and automation of the processes to, in some cases, an industrial level (Parker, 2005). For example, the Mediterranean fruit fly mass-rearing facility in El Pino, Guatemala, currently produces 2000 million sterile male insects per week but has the capacity to produce 3500 million/week (Fig. 3, Table 1) (Tween, 2002; IAEA, 2008).

As the scale of rearing increases, the selection pressure on the insect to adapt also increases and this can lead to behavioural changes in the colony that impact on the ability of the sterile insects to perform adequately in the field. The quality of insects for use in programmes has to be defined in terms of "fit for purpose". In the rearing facility, high productivity and survival are paramount, whereas in the field, the only quality aspects of relevance are the ability of the sterile males to find, mate with and effectively inseminate wild females (Lance et al., 2000; Vreysen, 2005). These differing quality requirements are not always readily attained, but the importance of the quality of the released insect in the field rather than performance in the rearing facility is receiving more and more attention (Simmons et al., 2010).

## 3.2 Producing Sterile Males

Successful use of sterile insects requires that the sterile males locate, mate with and effectively fertilise the wild females (Knipling, 1979). The basic reproductive biology of most insects would indicate that only males



**Fig. 3** (Upper, left) the Mediterranean fruit fly mass-rearing facility in El Pino, Guatemala, (lower, left) heat treatment of eggs to remove the female sex – each water bath contains 390 million

eggs, (upper, right) racks of cages with adult flies, and (lower, right) racks with larval rearing trays (photos courtesy of Gerald Franz, reproduced with permission)

need to be released for the sterile insect technique to be effective (Franz, 2005). What, if any, is the role of the released sterile females in the field? At best they can be considered neutral but at worst they may provide an opportunity for assortative mating with the released sterile males and so divert the males from seeking out and mating with wild females (Rendón et al., 2000, 2004).

Mathematical modelling has indicated that in species where there is a male-choice mating system (such as the screwworm), asymmetric mating due to colonisation processes (sterile males mate at random with wild and sterile females, whereas wild male screwworms mate preferably with wild females) requires a doubling of the amount of sterile insects released per unit surface area to get the same level of suppression as compared with a situation where there is random mating. The release of only sterile males of a species that has a female-choice mating system such as tropical fruit fly species - is more efficient in population suppression than the release of both sterile males and sterile females. In such species the number of multiple matings per male needs to be minimised because both sperm quality and quantity diminish with successive matings. In addition, sterile males distribute

themselves more widely when sterile females are absent (Vreysen et al., 2006a).

Whatever the situation in the field, the production and release of sterile females incur an economic cost to the programme as they have to be reared, marked, irradiated and released along with the males. Rearing and releasing insects is a major cost factor and the removal of females would increase effectiveness (McInnis et al., 1994; Hendrichs et al., 1995; Rendón et al., 2004) and deliver significant economic benefits (LaChance, 1979). For human disease vectors such as mosquitoes where only the female is the obligatory blood feeder, the removal of females from the release insects is essential (Lance and McInnis, 2005).

Very few sex-linked differences in insects have been discovered thus far, which might facilitate large-scale separation of males and females. Consequently, New World screwworm (Baumhover, 1966) and Lepidoptera (Bloem et al., 2005) sterile insect technique programmes have been implemented by releasing both sexes. Discrete differences between the sexes can be produced if certain mutations (e.g. pupal colour, resistance to insecticides, sensitivity to temperature) are linked to either sex (Robinson, 2002). In the case of the Mediterranean fruit fly, a genetic sexing

Insect/s	Name	Country	City	Operated by	Capacity/week
Ceratitis capitata	Infruitec Medfly Facilities	South Africa	Stellenbosch	Agricultural Research Council	5 million
	USDA Pacific Basin Area Research Center	United States of America	Honolulu	USDA/ARS	5 million
	Medfly facility	Tunisia	Tunis	National Center of Nuclear Sciences and Technologies (CNSTN)	10 million
	UOC-Fruit Flies	Greece	Heraklion, Crete	Biology Department	10 million
	Bio-Fly	Israel	Sde Eliyahu	Bio-Fly	25 million
	Centro Produccion Insectos Esteriles	Chile	ARICA	Servicio Agricola y Ganadero (SAG)	30 million
	Florida Department of Agriculture	United States of America	Gainesville	FDACS, DPI	50 million
	Biofábrica do Programa Madeira-Med Centro de Producción y Esterilización	Portugal	Camacha	Madeira Regional Governement	50 million
	Mosca de la Fruta La Molina - CPELM	Peru	Lima	Servicio Nacional de Sanidad Agraria	160 million
	Bioplanta KM8	Argentina	Mendoza	ISCAMEN	200 million
	Planta Moscamed, Metapa	Mexico	Metapa de Dominguez	Senasica/USDA	550 million
	El Pino	Guatemala	Santa Rosa	USDA/MOSCAMED	3000 million
Bactrocera tryoni	Queensland fruit fly production facility	Australia	Camden	New South Wales Agriculture	15 million
Bactrocera philippinensis	Philippine Fruit Fly Mass Rearing Facility	Philippines	Quezon City	Philippine Nuclear Research Institute	15 million
Bactrocera dorsalis	Irradiation Center for Agricultural Development	Thailand	Bangkok	Department of Agricultural Extension	30 million
Anastrepah ludens	Mexican Fruit Fly Rearing Facility	United States of America	Edinburg	USDA APHIS PPQ	160 million
Bactrocera cucurbitae	Okinawa Prefectural Agricultural Experiment Station	Japan	Okinawa	Okinawa Prefectural Government	200 million
Anastrepha ludens, A. obliqua	MOSCAFRUT	Mexico	Chiapas	Mexico Government	600 million
Bactrocera oleae	Bio-Fly	Israel	Sde Eliyahu	Bio-Fly	0.04 million
Pectinophora gossypiella	Pink Bollworm Rearing Facility	United States of America	Phoenix	USDA-CDFA	196 million
Cydia pomonella	Okanagan-Kootenay Sterile Insect Release Comisión México Americana para la	Canada	Osoyoos	Regional Districts (local government)	15 million
Cochliomyia hominivorax	Erradicación del Gusano Barrenador del Ganado.	Mexico	Chiapas	USDA-SAGARPA	500 million

 Table 1
 Selected rearing facilities and production capacity of various fruit flies, screwworm flies and moths

strain that carries two mutations (*temperature-sensitive lethal (tsl)* (Franz et al., 1994) and *white pupae (wp)* (Rössler, 1979)) is now routinely used in all sterile insect release programmes for this species, the global weekly production being 3500 million (Franz, 2005).

# 3.3 Inducing Sterility

A major obstacle facing the entomologists who pioneered the use of sterile insects was that they did not know of any means to induce sexual sterility in insects.

Although X-rays had been shown in the early part of the century to be able to sterilise insects (Runner, 1916) it was not until 1950 that, following communications between Müller and Knipling's group, the use of ionising radiation was considered as a potential technique for insect sterilisation (Bushland and Hopkins, 1951, 1953). Geneticists understood very well that ionising radiation induced sterility, i.e. dominant lethal mutations in germ cells of irradiated insects (Van Borstel, 1962). These mutations are caused by breakage of chromosomes in the germ cells and when sperm carrying these mutations fertilise eggs, the resulting embryo dies (LaChance et al., 1967). The word "sterility" suggests a single state or condition; however, ionising radiation induces levels of sterility, which are dependent on the dose of radiation used: the higher the dose, the higher the level of sterility (LaChance et al., 1967; Van der Vloedt et al., 1978; Vreysen, 1995; Vreysen et al., 1996). Ionising radiation not only induces dominant lethal mutations in the germ cells but also causes mutations in somatic cells, and in some species this impacts on the overall quality of the insect after radiation (Sakurai et al., 2000), expressed as the development of abnormalities, a reduction in lifespan, flight ability, mating propensity, etc. (Bakri et al., 2005a). In general, doses of ionising radiation have been used which lead to full sterility and this impacts negatively on insect quality, although recent publications (Toledo et al., 2004; Parker and Mehta, 2007) suggest that using a lower dose with less sterility but increased quality is more efficient at introducing sterility into the field population. Using a mathematical model, Klassen and Creech (1973) found that the rate of increase in a population to be suppressed determines the effectiveness of the combination of levels of male sterility and release ratios. When the rate of increase is low (e.g. five-fold), 90% male sterility can be effective at fairly low release ratios. By contrast, when the rate of increase is high (ten-fold or more) the level of sterility must be near 99% to induce a steep downward trend in the population (Klassen and Creech, 1973).

Sensitivity levels of insects to ionising radiation are affected by the level of oxygen present during irradiation (Economopoulos, 1977; Fisher, 1997). In an oxygen-reduced environment, the formation of free radicals during irradiation is reduced as compared with air, which lowers the damage induced by the ionising radiation. This entails the need for higher doses to obtain the same level of sterility (Bakri et al., 2005). This effect tends to be higher for somatic damage than for sterility and therefore, the use of hypoxia (a deficiency of oxygen reaching the tissues of the body) can be a strategy used to increase the competitiveness of an insect without compromising the level of sterility (Calkins and Parker, 2005; Lance and McInnis, 2005). Tephritid pupae are normally sealed in airtight plastic bags, with as little air as possible, and are left for about one hour before irradiation. During this period, the pupae exhaust most of the oxygen in the container. Saturating the atmosphere with helium or nitrogen before and during irradiation is another way of creating hypoxia (Ashraf et al., 1975; Hooper, 1989; Vreysen and Van der Vloedt, 1995).

Insect species vary in their sensitivity to the induction of sterility by ionising radiation, sometimes by an order of magnitude; for example, less than five Gy is required to sterilise Acrididae (Orthoptera) and Blaberidae (Dictyoptera), whereas doses above 200 Gy are in most cases required to sterilise pupae or adult Lepidoptera (Bakri et al., 2005a, b). However, in general insects can be divided into two groups depending on the type of chromosomes they possess. Some orders such as Lepidoptera and Hemiptera have holokinetic chromosomes (properties of the centromere are distributed over the entire chromosome), which are very radiation-resistant and high doses of ionising radiation are required to induce sterility. However, the majority of insect Orders have monokinetic chromosomes and are considerably more radiosensitive. The high radiation doses needed to sterilise Lepidoptera reduced quality and hence their usefulness in field programmes. However, Proverbs (1962) discovered quite serendipitously that if codling moths Cydia pomonella (L.) were irradiated with a lower dose, the females were sterilised but not the males and the progeny from these males, both males and females, were fully sterile. This phenomenon has been termed F<sub>1</sub> sterility or inherited sterility (North, 1967, 1975; LaChance, 1985).

Field releases of several lepidopteran pests such as the cabbage looper *Trichoplusia ni* (Hübner) (North and Holt, 1969), the corn earworm *Helicoverpa zea* (Boddie) (Carpenter and Gross, 1993), the gypsy moth *Lymantra dispar* (L.) (Mastro, 1993), the pink bollworm *Pectinophora gossypiella* (Saunders) (Staten, 1993; Walters, 2000) and the codling moth *C. pomonella* (Bloem et al., 2001) have demonstrated the efficacy of the inherited sterility.

#### 3.4 Sterile Insect Quality Management

The success of an area-wide integrated pest management programme that incorporates the release of sterile insects is dependent on many factors (Vreysen et al., 2007b), with adequate sterile insect quality being one of the most important. Released males have to intermingle rapidly with the wild population, locate the wild virgin females and be competitive with their wild counterparts in mating with wild females (Vreysen, 2005). The quality of the released insect can easily be impaired by aspects related to colonisation and mass-rearing processes (Parker, 2005), sterilisation with ionising radiation (Bakri et al., 2005a), and marking, handling, transport and release procedures (Calkins and Parker, 2005; Dowell et al., 2005). The quality of the reared insect has often mainly been assessed in the rearing facility, with little attention being given to its competitiveness once released in the target area (Vreysen, 2005). Quality management parameters in the mass-rearing facilities were more related to production processes (quantity) than to competitiveness (quality) in the field (Simmons et al., 2010). The sterile males' ability to seek out the native virgin females and to transfer the sterile sperm is a vital component of a sterile insect technique programme (Miyatake and Yamagishi, 1993). The monitoring in the field of the sterile males' sexual competitiveness, their mobility and dispersal capacity, and their dispersion (spatial distribution within the habitat) needs to be carried out rigorously, frequently and accurately to provide the programme managers with the data needed to make the necessary programme decisions (Vreysen, 2005).

## 3.5 Monitoring Insects

As in most pest control programmes, the effectiveness of the applied control tactics is determined by monitoring the target insect population before, during and after a particular treatment. When sterile insects are used it is also necessary to monitor this component of the insect population in the field and to be able to differentiate released insects from wild insects. In some programmes it is also possible to monitor directly the induced sterility in wild females (Vreysen et al., 2000). Monitoring adult density or abundance is carried out using traps, which sample the insect population and provide some estimates of relative abundance. Trap design, placement, efficiency and insect activity all interact to provide on a regular basis some data on the numerical fluctuations of the wild population. This direct assessment can be complemented by indirect methods, e.g. fruit sampling to assess larval infestation levels for fruit flies and parasite levels in the case of insect vectors of disease (Vreysen, 2005).

Sterile insects are generally marked in the rearing facility with a fluorescent dust before release to be able to distinguish them from the wild insects when caught in traps but also to be able to monitor their survival and dispersal in the field (Parker, 2005). The ratio of wild males to sterile males is a key parameter in determining progress in the programme. An increasing ratio in favour of sterile males to wild males indicates that progress is being made, and if this can be combined with direct measurements of sterility levels in wild females then very important estimates of the competitiveness of the sterile males can be obtained (Vreysen, 2005).

A key question in eradication programmes is establishing absence of the pest: absence of evidence is not evidence of absence. This question is compounded by trap efficiency, which may very well change when field populations are driven to very low absolute densities. This problem can to a large extent be solved by knowing the generation time of the population in the field, having some estimates of trap efficiency and accepting some level of probability (Barclay et al., 2005). A recent mathematical model calculates, given a certain amount of traps in an area, an estimate of the efficiency of the used traps, the number of trapping days to catch an insect with a certain statistically acceptable level of probability (Barclay and Hargrove, 2005). This user-friendly model (written in Excel) can be a great cost-saving factor, as it will not only give information of absence of the pest with a certain level of confidence, it will also provide guidance when to stop the releases (Clift and Meats, 2002, Vreysen, 2005).

In operational programmes that include the release of sterile males, the decision to stop releases is often influenced by political and financial motives. As an example, the dispersal of sterile New World screwworms in the Central American programme continued for 6-18 months after the last case had been detected (FAO, 1992; Wyss, 2000). In fruit fly programmes, it is standard procedure that releases continue for at least three fly generations after the last wild fly has been trapped (IAEA/FAO, 1997).

## 3.6 Using Sterile Insects

Although historically the use of sterile insects has been associated with the elimination of local pest populations, and although sterile insects always function in the field by introducing sterility into wild females, they can be deployed in several different ways depending on the ecological situation, the pest threat and the programme goals. Sterile insects have been used successfully in eradication, suppression, containment and prevention programmes (Hendrichs et al., 2005) (see boxes 1, 2, 3, 4). These options illustrate the flexibility with which sterile insects can be used to protect agricultural commodities from pests, be they endemic or exotic.

#### Box 2 Eradication

Eradication is defined as the application of phytosanitary measures to eliminate a pest from an area (FAO, 2005). This clearly implies the elimination of a local population, contrary to the definition used by the World Health Organization (WHO) where the term is restricted to global extinction (e.g. smallpox). The control efforts for the local eradication of a pest insect population are limited in time and cost, will eliminate all insecticide use in the target area eventually, and provide access to specific lucrative pestfree export markets. The rigorous quarantine setup needed to maintain the area pest-free is costly, but this also benefits quarantine procedures against other pest insects. An eradication strategy usually requires a high, but short-term investment, and long-term good monitoring networks (Hendrichs et al., 2005).

## Box 1 Suppression

The FAO (2005) defines suppression as the application of phytosanitary measures in an infested area to reduce pest populations. Using this strategy, sterile insects are deployed simply as a biological insecticide in order to maintain the pest population below an agreed and acceptable economic injury level. This strategy has the advantage of decreased pesticide use, lower investment needed for monitoring and no need for rigorous quarantine measures, but it requires preexport treatment (as part of a systems approach to develop Areas of Low Pest Prevalence), active participation of the growers, and the control effort has to be permanent (Hendrichs et al., 2005). This strategy has been applied successfully against the codling moth Cydia pomonella (L.) (Canada), the Mediterranean fruit fly Ceratitis capitata (Wiedemann) (Israel/Jordan, South Africa, Madeira, Spain and Tunisia), the oriental fruit fly Bactrocera dorsalis Hendel (Thailand), and the false codling moth Thaumatotibia leucotreta (Meyrick) (South Africa).

#### Box 3 Containment

Containment programmes are adopted in and around an infested area to prevent spread of the pest (FAO, 2005). Classical examples are (1) the containment programme of the pink bollworm *Pectinophora gossypiella* (Saunders) in the San Joaquin Valley, where sterile insect releases have, since 1968, prevented the establishment of the pink bollworm in the valley, protecting a cotton crop with an annual value of USD 1000 million (Bloem et al., 2005), and (2) the containment programme of the Mediterranean fruit fly Ceratitis capitata (Wiedemann) on the Mexico-Guatemala border, that has protected the horticultural industry in Mexico and the USA since 1983. It is estimated that for every dollar invested in this programme, there has been a return of USD 167 in terms of crop revenue and savings in control costs (Enkerlin, 2005).

#### Box 4 Prevention

Prevention is defined as the application of phytosanitary measures in and around a pest-free area to avoid the introduction of a pest. This strategy is the most desirable in terms of cost as it is always cheaper to prevent outbreaks than control them, as is exemplified by the preventive sterile release programme in California. Periodic outbreaks of the Mediterranean fruit fly Ceratitis capitata (Wiedemann) in California were previously controlled by unpopular aerial applications of Malathion. Outbreaks in 1980 and 1982 required more than USD 100 million to bring the situation under control. In 1994, a preventive release programme was started which releases 400 million sterile male fruit flies per week over the 2500-square-mile target area. The programme costs about USD 15 million/year, which compares favourably with the cost of USD 30 million that was spent annually to control the outbreaks (Vreysen et al., 2006b).

# 4 Ionising Radiation and Natural Enemies

During the past 120 years, more than 2000 species of exotic arthropods have been introduced on more than 5000 occasions into 196 countries for the control of arthropod pests. Many new exotic pests have established themselves in Europe, temporarily or permanently, as a result of increased acreage of protected grown crops. Biological control by augmentation or inundation is now a major component of pest control in protected crops in Europe (Loomans, 2006). The number of exotic natural enemies introduced and released has increased the last few decades, with about 90 species of natural enemies currently used and commercialised around Europe (EPPO, 2002). The increase in natural enemies shipped and released to control native and exotic pests may result in an increase in risks imposed on the environment, which has been the topic of several recent publications (e.g. Bigler et al., 2006). However, when appropriately applied, natural enemies are one of the most promising, environmentally-sound and effective tools for sustainable control of arthropod pests of agriculture, and biological control has been an accepted method of pest management for over 120 years in agriculture, forestry and rangeland areas and for medical and veterinary pests (van Lenteren et al., 2006; van Driesche et al., 2008).

In classical or inoculative biological control, nonindigenous biological control agents, usually selected from the suite of parasitoids, predators and diseases that co-evolved with the pest are introduced into the target area. One of the key concerns of this approach is the host specificity and host range of the natural enemies following release. There is always the possibility that introduced biological control agents "jump hosts" (Follet and Duan, 2000; Wajnberg et al., 2001), leading to criticisms (Hamilton, 2000; Louda et al., 2003) and ever more stringent environmental risk assessment protocols (van Lenteren et al., 2006). In cases where doubts remain about very promising natural enemies of weeds or insect pests, they can be radiationsterilised before release to enable a more definite assessment to be made under natural conditions of host specificity without any risk of permanent establishment (Hendrichs et al., 2009).

Ionising radiation has other roles to play in the area of augmentative or inundative biological control where natural enemies are mass-reared and released in very large numbers, often several times each season. Augmentative biological control faces challenges including the cost of production, quality control and quality assurance, shipping and regulation, and radiation can facilitate some of these issues (Hendrichs et al., 2009).

# 4.1 Ionising Radiation for Improved Rearing

Ionising radiation can sometimes suppress the immune response of the hosts, and thereby make older larval instars suitable for parasitoid development. A radiation-impaired immune system would also allow the production of natural enemies on easier to rear factitious or non-habitual hosts (Genchev et al., 2007) or on those that are more economical to mass-rear (Hamed et al., 2009). Certain complex, difficult to study physiological processes in the host (e.g. defence mechanisms, hormone metabolism), which are
not well understood in relation to improved rearing efficiency, can be selectively modified by ionising radiation, thereby facilitating the study of particular hostparasitoid interactions. Ionising radiation can arrest development of certain species of hosts and prey, thereby increasing their shelf life and allowing for storage and stockpiling (Celmer-Warda, 2004; Seth et al., 2009; Zapater et al., 2009). It can also delay normal insect development and thus may extend the time window for host parasitisation or modify the internal host environment to the benefit of the natural enemy (Cancino et al., 2009; Fatima et al., 2009; Hamed et al., 2009; Zapater et al., 2009). In most insect mass-rearing facilities, excess production of particular insect life stages or batches of sub-standard insects are normally discarded, which requires an investment for their disposal. These excess products may be processed instead (Nakashima et al., 1996) or irradiated, if still alive, to support the production of natural enemies (Cancino et al., 2009).

# 4.2 Ionising Radiation to Facilitate Handling, Shipment, Trade and Release

During mass-production of natural enemies, a certain proportion of the hosts or prey insects may not be parasitised or eaten and will therefore develop normally, which requires additional handling steps to remove these fertile individuals from the rearing process. This decreases efficiency in large-scale mass-rearing, and can be avoided by irradiating the hosts and prey to inhibit further development and prevent the need for costly separation procedures (Cancino et al., 2009).

# 4.3 Integrating the Sterile Insect Technique (SIT) or F<sub>1</sub> Sterility and Natural Enemies

The release of sterile or semi-sterile insects together with natural enemies has been known to have synergistic effects for population suppression when applied simultaneously (Knipling, 1992; Wong et al., 1992; Bloem et al., 1998). This synergy results from the sterile insects curtailing reproduction of the adult stage of the target pest, while natural enemies destroy one or more of its immature stages, including reproducing on the F<sub>1</sub> offspring in F<sub>1</sub> sterility releases. However, successful integration of these two control tactics is only possible if the natural enemies do not negatively impact the irradiated insects and their progeny more than the wild population, and if F<sub>1</sub> sterility does not negatively impact the efficacy and reproduction of natural enemies. The possibility of parasitisation on progeny from irradiated males and untreated females was demonstrated in studies in the laboratory, field cages and in the wild with irradiated and untreated (1) male Spodoptera exigua (Hubner) as a host for the braconid larval parasitoid Cotesia marginiventris (Cresson) (Carpenter et al., 1996), (2) male H. zea as a host for the tachinid Archytas marmoratus (Townsend) (Mannion et al., 1994), and (3) C. pomonella as a host for the egg parasitoids Trichogramma cacoeciae (Marchal) and Trichogramma nerudai (Pintureau and Gerding) (Botto and Glaz, 2010).

### **5** Protecting the Environment

The presence of sterile insects in the environment presents little or no threat (Nagel and Peveling, 2005). They are biodegradable, they act in a species-specific manner and they do not persist when releases are terminated. Consequently, the sterile insect technique is considered to be amongst the most environmentallyfriendly insect pest control tactics. However, prior to arriving in the environment they have to be reared, sterilised and released, all processes that impact the environment in some way, but probably in a similar way to the manufacture of other products for pest control. Large rearing facilities have to be built and they produce many tons of spent diet a day that has to be treated before being used for other purposes, and waste water management is a priority (IAEA, 2008). Bio-security is an issue where facilities find themselves in eradicated areas through the course of a successful programme, e.g. the New World screwworm mass-rearing facility in Tuxtla Gutièrrez in Chiapas, Mexico (Wyss, 2000). For insect sterilisation, conventional self-shielded irradiators (Fig. 4) are used and the dose received by the irradiated insects is quantified by standard dosimetry (Parker and Mehta, 2007).

Fig. 4 The three <sup>137</sup>Ce Husman gamma irradiators at the New World screwworm mass-rearing facility in Tuxtla Gutièrrez in Chiapas, Mexico

The transport and importation of this type of gamma cell is governed by international safety procedures and standards and its operation is under strict control by competent national authorities to ensure its safety and security. Releasing insects in large programmes is frequently carried out by aircraft, which again is well regulated and poses no environmental concerns. Many of these same concerns also apply to the mass-production of natural enemies, although biosecurity is probably less of a concern, except where pest species are used as hosts.

#### 6 Enhancing Sustainable Agriculture

Area-wide integrated pest management programmes using sterile insects and natural enemies are complex and management-intensive, and there is no guarantee of success (see Vreysen et al., 2007b). It is beyond doubt that the use of these biological control agents in area-wide integrated pest management programmes has resulted in high benefit-cost ratios (Mumford, 2005) and these programmes have played and continue to play a significant role in reducing the total amount of insecticides applied (Vreysen et al., 2006b).

The highest benefit-cost ratios are usually obtained in programmes that are operated to protect a high-value crop and its export market (e.g. export of fruits, vegetables, etc. to large markets of industrialised nations such as the EU, USA and Canada) or to protect an extensive livestock industry (e.g. cattle industry in southern USA). In those cases, it has been relatively easy to convince federal or regional governments, and farmers' associations to invest the necessary funds to operate these programmes as the benefits are very obvious. In the case of protecting subsistence farming in sub-Saharan Africa, it becomes much more difficult to find the funding for these programmes as there are no big industries or export markets to protect: what is protected are individual animals or crops, herds, and the livelihood of farmers. However, the very impressive programme that controlled the cassava mealy bug in large areas of Africa using a natural enemy showed the potential, even under very difficult conditions (Herren and Neuenschwander, 1991).

Insect pests cause both direct and some indirect losses, which translate into direct and indirect benefits when the pest is managed in an appropriate way. Direct benefits are fairly easy to account for and are measured in terms of increase in yield (e.g. number of fruits produced, milk and meat yield, reduced mortality, reduced abortions, increased longevity, etc.), improved quality of the produced commodities, and reduced production costs through more effective management of the pest (Enkerlin, 2005). Indirect benefits are more difficult to incorporate in the benefit-cost equations and can include an increase in export volumes, increase in yield through reduced secondary pest outbreaks, savings in medical costs as a result of exposure to insecticides, reduced legal costs arising from damage to private and public property due to insecticide misuse, increased human health due to reduced insecticide residues in food commodities, new jobs created, better human nutrition, etc. (Enkerlin, 2005).

The following examples provide ample evidence of the significant contribution area-wide integrated pest management has made to sustainable agriculture.



### 6.1 New World Screwworm in the Americas

Using an area-wide integrated pest management approach and sterile insects, the New World screwworm was eradicated from the USA during the period 1957-1966 (Knipling, 1960) although there were several reintroductions, with the last six cases in Texas in 1982 (Jim Novy, unpublished report to the FAO). Mexico was declared free of the New World screwworm in 1991, and the remainder of Central America was declared New World screwworm-free over a period of 12 years (from 1988 to 2000) (Vargas-Terán, 2005). The annual direct producers' benefits from the eradication of screwworm were estimated at USD 896 million for the USA, USD 328 million for Mexico and USD 87.8 million for Central America (Wyss, 2000). The benefits were derived from decreases in death losses, decreases in veterinary services, veterinary medicines, insecticides and inspections, and increases in meat and milk production. These effects have, however, an indirect impact on the general economy of each country and the multiplier effect is estimated at 3 (Wyss, 2000). Since the first sterile New World screwworms were released on Curaçao, an estimated USD 1000 million has been spent on the eradication campaign over 50 years. The annual direct benefits that accrue every year from this programme are therefore equal to or more than the total project cost over five decades. Whereas in 1934 screwworm infestations killed 1.3 million cattle in the south-eastern USA (Dove, 1937) and in 1935, in Texas alone, 230 000 screwworm cases were detected (FAO, 1992), the screwworm saga has been erased from the memories of today's US livestock owners.

### 6.2 Chile's National Fruit Fly Programme

Export of fruit and vegetables is an important component of the economy of Chile. Chile's most important trading partners are the USA and Japan, who impose stringent import conditions: the produce needs to be produced in a fruit fly-free area. Only the northern province of Arica was infested with the Mediterranean fruit fly, and following the use of bait sprays and sterile insects, Chile was declared a fruit fly-free country in 1995 (MAG/SAG, 1995). Since then, Chile's National Fruit Fly programme has been able to prevent the introduction and the establishment of fruit flies of economic importance. The programme uses two main strategic activities, i.e. (1) a preventive programme based on effective quarantine measures and an extensive trapping network, and (2) in collaboration with Peru and Argentina, a containment programme in Arica province using twice-weekly releases of 50 million sterile males (Gonzalez and Troncoso, 2007). Since Chile was declared a fruit fly-free country, annual fruit exports have grown to two million tonnes of mainly grapes, apples, stone fruits, kiwis and avocados with an estimated value of USD 1600 million. The annual export of fruit from the area that could harbour the Mediterranean fruit fly amounts to USD 250 million. However, the Government of Chile spends only USD 4 million each year to keep the country free of fruit flies (Enkerlin, 2005).

### 6.3 Melon Fly in Okinawa

The melon fly B. cucurbitae is one of the most destructive pests of cucurbit crops. This pest invaded the Okinawa Islands of Japan from 1910 to 1970, thereby preventing the export of 40 important vegetables and fruits to fruit fly-free areas, such as the main islands of Japan. In 1979, an area-wide integrated pest management programme was initiated that integrated the use of the male annihilation technique with the release of sterile insects. All islands were declared free of melon fly in 1993 and this status is maintained through a continuous detection programme, rigorous quarantine measures and preventive releases on the island closest to Taiwan that is still infested. As a result of the programme, between 1990 and 2000, melon production increased 2.3 times (from 2720 to 6220 tonnes) and the production of mango 4.6 times (from 278 to 1290 tonnes), or an equivalent of USD 335 million. A total of USD 172 million was invested for the main eradication campaign between 1982 and 1991. However, six years after achieving eradication, the breakeven point had been reached, and between 1997 and 2000, revenues from the sale of commodities to mainland Japan totalled USD 169 million. For the same period, only USD 31 million was required to keep the islands free of new invasions (Enkerlin, 2005).

#### 6.4 Cassava Mealy Bug in Africa

Although the theoretical basis for the combined use of sterile insects and natural enemies was already established by Knipling (1992) and numerous laboratory, semi-field and field experiments have indicated the synergistic benefits (e.g. Wong et al., 1992; Bloem et al., 1998), the principle has not found a practical follow-up "as yet" on a large operational scale. This is remarkable as both sterile insects (Dyck et al., 2005) and natural enemies (Hokkanen and Lynch, 1996) have been used successfully to tackle major insect pests. One of the most impressive classical biological control programmes was undoubtedly the one directed against the cassava mealy bug Phenacoccus manhoti Mat.-Ferr. (Homoptera: Pseudococcidae) in Africa. The mealy bug was first detected in Zaire and Congo in the early 1970s (Herren and Neuenschwander, 1991) and spread the following 10 years over the entire cassava belt of Africa, with the exception of Madagascar, causing root yield losses of up to 84% (Herren et al., 1987). A parasitoid wasp Apoanagyrus (Epidinocarsis) lopesi (DeSantis), originating from the mealy bug's native home Paraguay, was selected as the prime candidate for a release programme. Aerial releases of the parasitoid wasp were initiated in 1981 and by 1991 the biological control agent had been spread to 26 African countries covering 3 million km<sup>2</sup> (Neuenschwander, 1993), bringing the pest under control in 95% of all the cassava fields (Herren and Neuenschwander, 1991). An economic analysis that tabulated the benefits and costs over a period of 40 years (1974-2013) indicated a benefit-cost ratio of the programme of 200 when cassava was listed at world market prices and of about 370-740 when inter-African prices were considered (Zeddies et al., 2001).

# 6.5 The Private Sector Enters the Fruit Fly Industry in Israel

Suppression of the Mediterranean fruit fly in the Arava/Araba Valley (Israel/Jordan) was initiated in 1998. Initially, the programme imported 15 million sterile male pupae weekly from the El Pino Moscamed mass-rearing facility in Guatemala. After September 11, 2001, airfreight regulations changed drastically, which significantly increased the duration of the pupal shipments, with increased pupal mortality and a general decline in fly quality as a consequence. As a result, Bio-Bee, an Israeli company that massproduces and markets biological control agents and bumblebees, conducted in 2003 a study to assess the economic feasibility of establishing a rearing facility. A new company named Bio-Fly was established in the Kibbutz Sde Eliyahu and the rearing facility was inaugurated in March 2005. The facility currently produces 20-25 million sterile male flies every week (Franz, 2005), although its maximum production capacity is estimated at 50 million sterile males per week using the existing infrastructure. Bio-Fly delivers sterile male flies to projects in the Golan Heights (800 ha. of deciduous fruits), the Sea of Galilee (800 ha. of mangos and citrus), to Beit-Shean (300 ha. of citrus), to Lachish Tali (600 ha. of grapes), to the West Negev (200 ha. of citrus), and to the Arava-Araba Valley (4000 ha. of commercial plots and backyards) (Dael Levy, General Manager, Bio-Fly, pers. commun.). Whereas before 1998 the Arava Valley was not able to export any agricultural commodity to the USA, export of bell peppers and tomatoes to the USA had increased to USD 500 000 in 1999 and to USD 50 million in 2003-2004.

#### 6.6 Codling Moth in Canada

The codling moth C. pomonella is the most important pest of pears and apples in British Columbia, Canada (Bloem and Bloem, 2000). A pilot programme, conducted from 1976 to 1978 in the Similkameen Valley, BC, demonstrated that a combination of suppressive action with chemicals and the release of sterile moths could eliminate codling moth (Proverbs et al., 1982). Operational activities were initiated in the Okanagan Valley in 1995 and since then, the apparent density of the wild codling moth population in the first intervention zone (as revealed through adult moth catches in pheromone traps) was reduced from an average of 2.5-13 moths/trap/week to an average of 0.08 moths/trap/week in 2000 (Bloem et al., 2005). In the first zone, the proportion of orchards with no detectable codling moth damage increased from 42% in 1995 to 91% in 2000, while the amount of organophosphate insecticides purchased decreased from 18 903 kg in 1991 to 3403 kg in 2001 (Bloem et al., 2005).

#### 7 The Future

There are several areas where the sterile insect technique could be improved and made more efficient. The quality of released insects and hence, their ability to locate virgin females and transfer the sperm is of prime importance for success of the technique. The use of field-based protocols that rigorously assess the impact of changes in the rearing and handling on the quality of the insect should be standard procedure (Vreysen, 2005). One can compensate for low insect quality to a certain extent by releasing a larger number of insects, but there are limits to this both economically and biologically, i.e. below a certain quality factor, further increases will simply not be effective. Genetic studies need to shed light on the amount of insects needed to initiate insect colonies with adequate genetic variability and how this variability can be maintained throughout the rearing process. Insects adapt rapidly to artificial conditions and successful colonisation results in the selection for adapted genotypes that can negatively impact on their competitiveness. Strain replacement at regular intervals has been used in the past to compensate for this but is logistically very demanding. The filter rearing system concept, where a mother colony is maintained under relaxed, more natural conditions, has shown great potential for some fruit fly species. The mother colony is checked at every generation and unwanted individuals are removed. Eggs from the mother colony are used to produce a large colony from which insects are taken for release following 3-4 generations of amplification. This strategy not only reduced selection pressure but can also maintain the integrity of genetic sexing strains (Fisher and Cáceres, 2000; Cáceres et al., 2004). Other areas that have great potential to increase the efficiency of the sterile insect technique are improvement of artificial diets through the inclusion of micro-organisms that contribute nutrients, the addition of hormonal, nutritional, semio-chemical supplements to enhance sterile male performance, increased fitness through hybrid strains, increased shelf life of certain life stages that would make sterile insects available in large numbers when needed in the season, improved shipping procedures to reduce negative effects of transport, reducing the negative impact of ionising radiation, improved release procedures, monitoring techniques, etc. Modern biotechnology will most likely also contribute to increased efficiency and programmes could benefit from

transgenic techniques in terms of insect marking, genetic sexing, molecular sterilisation and disease refractoriness (Robinson and Hendrichs, 2005).

### 8 Conclusions

The suppression, containment, prevention or eradication of pest insects will in most cases result in more sustainable pest control when the various control tactics are applied on an area-wide basis, i.e. against an entire pest population within a delimited geographical area. Both sterile insects and natural enemies have been used very successfully in the past to control important insect pests. Although both control tactics are complementary and in some circumstances even synergistic, their combined use has not yet found a practical follow-up. Ionising radiation is not only a very suitable, effective and safe way to produce sterile insects, it can also be used in many ways to increase the usefulness of natural enemies for insect control. Both natural enemies and sterile insects, when appropriately used, have minimal negative effects on the environment. There are numerous examples of past and ongoing operational programmes that have successfully deployed sterile insects or natural enemies that have resulted in enormous economic benefits and in many cases have opened up very lucrative export markets of agricultural commodities. There is great potential to use these control tactics and strategic approaches against major insect pests in the future and the potential for economic growth and enhanced agriculture could motivate farmers to apply these approaches more in an area-wide context.

#### References

- Animal and Plant Health Inspection Service/United States Department of Agriculture (APHIS/USDA) (2001) Screwworm programme review, Final Report.
- Ashraf M., Chatha N., Ohinata K., Harris E.J. (1975) Melon flies: dosage-response and sexual competitiveness after treatment with gamma irradiation in a nitrogen atmosphere, J. Econ. Entomol. 68, 838–840.
- Bakri A., Metha K., Lance D.R. (2005a) Sterilizing insects with ionizing radiation, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and

practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 233–268.

- Bakri A., Heather J., Hendrichs J., Ferris I. (2005b) Fifty years of radiation biology in entomology: lessons learned from IDIDAS, Ann. Entomol. Soc. Am. 98, 1–12.
- Barclay H.J., Hargrove J.W. (2005) Probability models to facilitate a declaration of pest-free status, with special reference to tsetse (Diptera: Glossinidae), Bull. Entomol. Res. 95, 1–11.
- Barclay H.J., Hargrove J.W., Clift A., Meats A. (2005) Procedures for declaring pest free status, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 363–386.
- Barnes B.N., Eyles D.K., Franz G. (2004) South Africa's fruit fly SIT project – the Hex River Valley pilot project and beyond, in: Barnes B.N. (Ed.), Proceedings of the 6th International Symposium on Fruit Flies of Economic Importance, 6–10 May 2002, Stellenbosch, South Africa, Isteg Scientific Publications, Irene, South Africa, pp. 131–141.
- Baumhover A.H. (1966) Eradication of the screwworm fly. An agent of myiasis, J. Am. Med. Assoc. 196, 240–248.
- Baumhover A.H., Graham A.J., Bitter B.A., Hopkins D.E., New W.D., Dudley F.H., Bushland R.C. (1955) Screwworm control through release of sterilized flies, J. Econ. Entomol. 48, 462–466.
- Bigler F., Babendreier D., Kuhlmann U. (2006) Environmental risks of invertebrate biological control agents: methods and risk assessment, CABI publishing, Wallingford, UK.
- Bloem K.A., Bloem S. (2000) SIT for codling moth eradication in British Columbia, Canada, in: Tan K.H. (Ed.), Area-wide Control of Fruit Flies and Other Insect Pests, Proceedings of an International Conference on Area-wide Control of Insect Pests and the 5th International Symposium on Fruit Flies of Economic Importance, 28 May–5 June 1998, Penang, Malaysia, Penerbit Universiti Sains, Malaysia, pp. 207–214.
- Bloem K.A., Bloem S., Carpenter J.E. (2005) Impact of moth suppression/eradication programmes using the sterile insect technique or inherited sterility, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 677–700.
- Bloem S., Bloem K.A., Carpenter J.E., Calkins C.O. (2001) Season-long releases of partially sterile males for control of codling moth *Cydia pomonella* (Lepidoptera: Tortricidae), in Washington apples, Environ. Entomol. 30, 763–769.
- Bloem S., Bloem K., Knight A.L. (1998) Oviposition by sterile codling moths, *Cydia pomonella* (Lepidoptera: Tortricidae) and control of wild populations with combined releases of sterile moths and egg parasitoids, J. Entomol. Soc. Brit. Columbia 95, 99–109.
- Botto E., Glaz P. (2010) Potential for controlling codling moth *Cydia pomonella* (Linnaeus) (Lepidoptera: Tortricidae) in Argentina using the sterile insect technique and egg parasitoids, J. Appl. Entomol. 134, 251–260.
- Brown H.E. (1984) Mass production of screwworm flies, *Cochliomyia hominivorax*, in: King E.G., Leppla N. (Eds.), Advances and challenges in insect rearing, USDA-ARS, New Orleans, USA, pp. 193–199.

- Bushland R.C., Hopkins D.E. (1951) Experiments with screwworm flies sterilized by X-rays, J. Econ. Entomol. 44, 725–731.
- Bushland R.C., Hopkins D.E. (1953) Sterilization of screwworm flies with X-rays and gamma rays, J. Econ. Entomol. 46, 648–656.
- Cáceres C., Cayol J.P., Enkerlin W., Franz G., Hendrichs J., Robinson A.S. (2004) Comparison of Mediterranean fruit fly (*Ceratitis capitata*) (Diptera: Tephritidae) bisexual and genetic sexing strains: development, evaluation and economics, in: Barnes B.N. (Ed.), Proceedings of the 6th International Symposium on Fruit Flies of Economic Importance, Isteg Scientific Publications, Irene, South Africa, pp. 367–381.
- Calkins C.O., Parker A.G. (2005) Sterile insect quality, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 269–296.
- Cancino J., Ruíz L., López P., Sivinski J. (2009) The suitability of Anastrepha spp. and Ceratitis capitata (Diptera: Tephritidae) larvae as hosts of Diachasmimorpha longicaudata and Diachasmimorpha tryoni (Hymenoptera: Braconidae): Effects of host age and radiation dose and implications for quality control in mass rearing, Bio. Sci. Tech. 19, S1, 81–94.
- Cardé R.T., Minks A.K. (1995) Control of moth pests by mating disruption: successes and constraints, Annu. Rev. Entomol. 40, 559–585.
- Carpenter J.E., Gross H.R. (1993) Suppression of feral *Helicoverpa zea* (Lepidoptera: Noctuidae) populations following infusion of inherited sterility from released substerile males, Environ. Entomol. 22, 1084–1091.
- Carpenter J.E., Bloem S., Marec F. (2005) Inherited sterility in insects, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in areawide integrated pest management. Springer, Dordrecht, The Netherlands, pp. 115–146.
- Carpenter J.E., Hidryani E., Sheehan W. (1996) Compatibility of F<sub>1</sub> sterility and a parasitoid, *Cotesia marginiventris* (Hymenoptera: Braconidae), for managing *Spodoptera exigua* (Lepidoptera: Noctuidae): acceptability and suitability of host, Fla. Entomol. 79, 289–295.
- Celmer-Warda K. (2004) Preliminary studies on suitability and acceptability of irradiated host-larvae (*Plodia interpunctuella*) by larval parasitoids *Venturia canescens* (Gravenhorst), Ann. Warshaw Agr. Univ-SGGW, Horticulture, Landscape Architecture 25, 67–73.
- Clift A., Meats A. (2002) When does zero catch in a male lure mean no tephritid flies in the area? in: Barnes B.N. (Ed.), Proceedings of the 6th International Symposium on Fruit Flies of Economic Importance, 6-10 May 2002, Stellenbosch, South Africa, Isteg Scientific Publications, Irene, South Africa, pp. 183–188.
- Courschee R.J. (1960) Some aspects of the application of insecticides, Annu. Rev. Entomol. 5, 327–352.
- Dame D.A. (1970) Control by sterilization of *Glossina*, in: Mulligan H.W. (Ed.), The African trypanosomiasis, Allen and Unwin Ltd, London, UK, pp. 533–542.
- Dove W.E. (1937) Myiasis of man, J. Econ. Entomol. 30, 29-39.
- Dowell R.V., Worley J., Gomes P.J. (2005) Sterile insect supply, emergence and release, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and

practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 297–324.

- Dyck V.A., Hendrichs J., Robinson A.S. (2005) Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands.
- Economopoulos A.P. (1977) Gamma-ray sterilization of *Dacus* oleae (Gmelin). Effect of nitrogen on the competitiveness of irradiated males, Z. Ang. Entomol. 83, 86–95.
- Enkerlin W.R. (2005) Impact of fruit fly control programmes using the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 651–676.
- European and Mediterranean Plant Protection Organization (EPPO) (2000) Safe use of biological control: import and release of exotic biological control agents, EPPO Standard PM6/2(1), http://www.eppo.org.
- European and Mediterranean Plant Protection Organization (EPPO) (2002), http://www.eppo.org.
- Fatima B., Ahmad N., Memon R.M., Bux M., Ahmad Q. (2009) Enhancing biological control of sugarcane shoot borer, *Chilo infuscatellus* (Lepidoptera: Pyralidae), through use of radiation to improve laboratory rearing and field augmentation of egg and larval parasitoids, Bio. Sci. Tech. 19, S1, 277–290.
- Food and Agriculture Organization of the United Nations (FAO) (1992) The New World screwworm eradication programme, North Africa (1988–1992), FAO, Rome, Italy.
- Food and Agriculture Organization of the United Nations (FAO) (1995) Glossary of phytosanitary terms, ISPM, Publication number 5, IPPC, FAO, Rome, Italy.
- Food and Agriculture Organization of the United Nations (FAO) (2005) International standards for phytosanitary measures, Guidelines for the export, shipment, import and release of biological control agents and other beneficial organisms (revision adopted at the 7th session of the ICPM, April 2005), Publication No. 3, Secretariat of the International Plant Protection Convention, FAO, Rome, Italy.
- Food and Agriculture Organization of the United Nations (FAO) (2008) FAOSTAT, http://www.fao.org/faostat/foodsecurity/ index\_en.htm.
- Fisher K. (1997) Irradiation effects in air and in nitrogen on Mediterranean fruit fly (Diptera: Tephritidae) pupae in Western Australia, J. Econ. Entomol. 90, 1609–1614.
- Fisher K., Cáceres C. (2000) A filter rearing system for mass reared medfly, in: Tan K.H. (Ed.), Area-Wide Control of Fruit Flies and Other Insect Pests, Proceedings of an International Conference on Area-Wide Control of Insect Pests, and the 5th International Symposium on Fruit Flies of Economic Importance, 28 May–5 June 1998, Penang, Malaysia, Penerbit Universiti Sains Malaysia, Pulau Pinang, Malaysia, pp. 543–550.
- Follet P.A., Duan J.J. (2000) Non-target effects of biological control, Kluwer Academic Publishers, Dordrecht, The Netherlands.
- Ford J., Nash T.A.M., Welch J.R. (1970) Control by clearing of vegetation, in: Mulligan H.W. (Ed.), The African trypanosomiasis, Allen and Unwin Ltd, London, UK, pp. 543–556.
- Franz G. (2005) Genetic sexing strains in Mediterranean fruit fly, an example for other species amenable to large-scale rearing for the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and

practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 427–251.

- Franz G., Gencheva E., Kerremans P. (1994) Improved stability of genetic sex-separation strains for the Mediterranean fruit fly, *Ceratitis capitata*, Genome 37, 72–82.
- Genchev N.P., Michelva-Dimitrova R.Y., Kozhuharova M.V. (2007) Use of gamma radiation for suppression of the hemocytic immune response in larvae of *Galleria mellonella* (Lepidoptera) against *Venturia canescens* (Hymenoptera), J. Balkan Ecol. 10, 411–419.
- Gonzalez J., Troncoso P. (2007) The fruit fly exclusion programme in Chile, in: Vreysen M.J.B., Robinson A.S., Hendrichs J. (Eds.), Area-wide control of insect pests. From research to field implementation, Springer, Dordrecht, The Netherlands, pp. 614–651.
- Hamed M., Nadeem S., Riaz A. (2009) Use of gamma radiation for improving the mass-production of *Trichogramma chilonis* and *Chrysoperla carnea*, Bio. Sci. Tech. 19, S1, 43–48.
- Hamilton G. (2000) When good bugs turn bad, New Scientist, 30–33.
- Hendrichs J., Franz G., Rendón P. (1995) Increased effectiveness and applicability of the sterile insect technique through maleonly releases for control of Mediterranean fruit flies during fruiting seasons, J. Appl. Entomol. 119, 371–377.
- Hendrichs J., Kenmore P., Robinson A.S., Vreysen M.J.B. (2007) Area-wide integrated pest management (AW-IPM): principles, practice and prospects, in: Vreysen M.J.B., Robinson A.S., Hendrichs J. (Eds.), Area-wide control of insect pests. From research to field implementation, Springer, Dordrecht, The Netherlands, pp. 3–33.
- Hendrichs J., Vreysen M.J.B., Enkerlin W.R., Cayol J.P. (2005) Strategic options in using sterile insects for area-wide integrated pest management, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 563–600.
- Hendrichs J., Bloem K.A., Hoch G., Carpenter J.E., Greany P., Robinson A.S. (2009) Improving the cost-effectiveness, trade and safety of biological control for agricultural insect pests using nuclear techniques, Bio. Sci. Tech. 19, 3–22.
- Herren H.R., Neuenschwander P., Hennesey R.D., Hammond W.N.O. (1987) Introduction and dispersal of *Epidinocarsis lopezi* (Hym., Encyrtidae), an exotic parasitoid of the cassava mealybug, *Phenacoccus manuhoti* (Hom., Pseudococcidae), Africa, Agric. Ecosyst. Environ. 19, 131–144.
- Herren H.R., Neuenschwander P. (1991) Biological control of cassava pests in Africa, Annu. Rev. Entomol. 36, 257–283.
- Hokkanen H.M.T., Lynch J.M. (1996) Biological control benefits and risks, Cambridge University Press, UK.
- Hooper G.H.S. (1989) The effect of ionizing radiation on reproduction, in: Robinson A.S., Hooper G. (Eds.), World crop pests, Vol. 3A, Fruit flies, their biology, natural enemies and control, Elsevier, Amsterdam, The Netherlands, pp. 153–164.
- International Atomic Energy Agency (IAEA) (2008) Model business plan for a sterile insect production facility, 978-92-0-110007-8, IAEA, Vienna, Austria.
- International Atomic Energy Agency/ Food and Agriculture Organization of the United Nations (IAEA/FAO) (1997) Control of the Mediterranean fruit fly in the Near East region

using the sterile insect technique, STI/PUB/1020, IAEA, Vienna, Austria.

- Klassen W. (2005) Area-wide integrated pest management and the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique, Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 39–68.
- Klassen W., Creech J.P. (1973) Population suppression with dominant and conditional lethal mutations; some important considerations and approaches, in: Computer models and application for the sterile male technique, IAEA, Vienna, Austria, pp. 65–79.
- Knipling E.F. (1955) Possibilities of insect population control through the use of sexually sterile males, J. Econ. Entomol. 48, 443–448.
- Knipling E.F. (1959) The sterile male method of population control, Science 130, 902–904.
- Knipling E.F. (1960) The eradication of the screwworm fly, Sci. Am. 203, 54.
- Knipling E.F. (1966) Some basic principles in insect population suppression, Bull. Entomol. Soc. Am. 12, 7–15.
- Knipling E.F. (1972) Entomology and the management of man's environment, J. Aust. Entomol. Soc. 11, 153–167.
- Knipling E.F. (1979) The basic principles of insect population suppression and management, Agriculture Handbook Number 512, SEA, USDA, Washington DC, USA.
- Knipling E.F. (1992) Principles of insect parasitism analyzed from new perspectives: practical implications for regulating insect populations by biological means, USDA Agriculture Handbook No. 693, USA.
- Koul O., Cuperus G., Elliott N. (2008) Areawide pest management. Theory and implementation, Cab International, UK.
- Kovaleski A., Mumford J. (2007) Pulling out the evil by the root; the codling moth *Cydia pomonella* eradication programme in Brazil, in: Vreysen M.J.B., Robinson A.S., Hendrichs J. (Eds.), Area-wide control of insect pests. From research to field implementation, Springer, Dordrecht, The Netherlands, pp. 581–590.
- Koyama T., Kakinohana H., Miyatake T. (2004) Eradication of the melon fly *Bactrocera cucurbitae* in Japan: importance of behaviour, ecology, genetics and evolution, Annu. Rev. Entomol. 49, 331–349.
- Krafsur E. (1998) Sterile insect technique for suppressing and eradicating insects; 50 years and counting, J. Agric. Entomol. 15, 303–317.
- Kuznetsova V.G., Petropavlovskaia M.B. (1976) Behaviour of holokinetic chromosomes in the spermatogenesis of bugs, Tsitologiia 18, 702–711.
- LaChance L.E. (1979) Genetic strategies affecting the success and economy of sterile insect release method, in: Hoy M.A., McKelvey J.J. Jr. (Eds.), Genetics in relation to insect management, Rockefeller Foundation, NY, WA, USA, pp. 8–18.
- LaChance L.E. (1985) Genetic methods for the control of lepidopteran species: status and potential, ARS-28, USDA/ARS, Washington DC, USA.
- LaChance L.E., Schmidt C.H., Bushland R.C. (1967) Radiation induced sterilization, in: Kilgore W.W., Doutt R.L. (Eds.), Pest control: biological, physical and selected chemical methods, Academic Press, New York, USA, pp. 147–196.
- Lance D.R., McInnis D.O. (2005) Biological basis of the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S.

(Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 69–94.

- Lance D.R., McInnis D.O., Rendón P., Jackson C.G. (2000) Courtship among sterile and wild *Ceratitis capitata* (Diptera: Tephritidae) in field cages in Hawaii and Guatemala, Ann. Entomol. Soc. Am. 93, 1179–1185.
- Lindquist D.A., Abusowa M., Hall M.J.R. (1992) The New World screwworm in Libya: review of its introduction and eradication, Med. Vet. Entomol. 6, 2–8.
- Loomans A.J.M. (2006) Regulation of invertebrate biological control agents in Europe: review and recommendations in its pursuit of a harmonised regulatory system, Report of REBECA, http://www.rebeca-net.de/downloads/.
- Louda S.M., Pemberton R.W., Johnson M.T., Follett P.A. (2003) Non-target effects: the Achilles' heel of biological control? Annu. Rev. Entomol. 48, 365–396.
- Ministerio de Agricultura/Servicio Agricola y Ganadero (MAG/SAG) (1995) Chile; a medfly-free country, Pamphlet, Government of Chile, Santiago, Chile.
- Mannion C.M., Carpenter J.E., Gross H.R. (1994) Potential of the combined use of inherited sterility and a parasitoid, *Archytas marmoratus* (Diptera: Tachinidae) for managing *Helicoverpa zea* (Lepidoptera: Noctuidae), Environ. Entomol. 23, 41–46.
- Marroquin R. (1985) Mass production of screwworms in Mexico, in: Symposium on eradication of the screwworm from the United States and Mexico, Proceedings of a Symposium Presented at the National Meeting of the Entomological Society of America, 9–13 December 1984, San Antonio, Texas, Entomological Society of America, MD, USA, pp. 31–36.
- Mastro V.C. (1993) Gypsy moth F<sub>1</sub> sterility program: current status, in: Radiation Induced F<sub>1</sub> Sterility in Lepidoptera for Area-Wide Control, Proceedings of the Final Research Coordination Meeting, Joint FAO/IAEA Division of Nuclear Techniques in Food and Agriculture, 9–13 September 1991, Phoenix, AZ, USA, STI/PUB/929. IAEA, Vienna, Austria, pp. 125–129.
- McInnis D.O., Tam S.Y.T., Grace C., Miyashita D. (1994) Population suppression and sterility rates induced by variable sexratio, sterile insect releases of *Ceratitis capitata* (Diptera: Tephritidae) in Hawaii, Ann. Entomol. Soc. Am. 87, 231–240.
- Meyer N.L. (1994) History of the Mexico-United States screwworm eradication program, Vantage Press, New York, USA.
- Miyatake T., Yamagishi M. (1993) Active quality control in mass reared melon flies. Quantitative genetic aspects, in: Management of Insect Pests: Nuclear and Related Molecular and Genetic Techniques, Proceedings of an IAEA/FAO International Symposium, 19–23 October 1992, Vienna, Austria. STI/PUB/909, IAEA, Vienna, Austria, pp. 201–213.
- Mumford J.D. (2005) Application of benefit/cost analysis to insect pest control using the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 481–498.
- Nagel P., Peveling R. (2005) Environment and the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique, Principles and practice in

area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 499–524.

- Nakashima Y., Hirose Y., Kinjo K. (1996) Rearing Orius sauteri (Poppius) on diet of freeze-dried larval powder of melon fly, Bactrocera cucurbitae Coquillett, Jap. J. Appl. Entomol. Zool. 40, 80–82.
- North American Plant Protection Organization (NAPPO) (2008) http://www.nappo.org/newsletter/2008/September.
- Neuenschwander P. (1993) Human interactions in classical biological control of cassava and mango mealy bugs on subsistence farms in tropical Africa, in: Altieri M.A. (Ed.), Crop protection strategies for subsistence farmers, IT Publications, London, Westview Press, San Francisco, USA, pp. 143–177.
- North D.T. (1967) Radiation-induced male sterility exhibited in the P<sub>1</sub> and F<sub>1</sub> generations in Lepidoptera, Rad. Res. 31, 615.
- North D.T. (1975) Inherited sterility in Lepidoptera, Annu. Rev. Entomol. 20, 167–182.
- North D.T., Holt G.G. (1969) Population suppression by transmission of inherited sterility to progeny of irradiated cabbage looper, *Trichoplusia ni*, Can. Entomol. 101, 513–520.
- Parker A.G. (2005) Mass-rearing for sterile insect release, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 209–232.
- Parker A.G., Mehta K. (2007) Sterile insect technique: a model for dose optimization for improved sterile insect quality, Fla. Entomol. 90, 88–95.
- Pimentel D. (2007) Area-wide pest management: environmental, economic and food issues, in: Vreysen M.J.B., Robinson A.S., Hendrichs J. (Eds.), Area-wide control of insect pests. From research to field implementation, Springer, Dordrecht, The Netherlands, pp. 35–47.
- Proverbs M.D. (1962) Progress on the use of induced sexual sterility for the control of codling moth *Carpocapsa pomonella* (L.) (Lepidoptera: Olethreutidae), Proc. Entomol. Soc. Ontario 92, 5–11.
- Proverbs M.D., Newton J.R., Campbell C.J. (1982) Codling moth: a pilot programme of control by sterile insect release in British Columbia, Can. Entomol. 114, 363-376.
- Rendón P., McInnis D., Lance D., Stewart J. (2000) Comparison of medfly-male only and bisexual releases in large scale field trials, in: Tan K.H. (Ed.), Area-Wide Control of Fruit Flies and Other Insect Pests, Proceedings of an International Conference on Area-Wide Control of Insect Pests, and the 5th International Symposium on Fruit Flies of Economic Importance, 28 May–5 June 1998, Penang, Malaysia, Penerbit Universiti Sains Malaysia, Pulau Pinang, Malaysia, pp. 517–525.
- Rendón P., McInnis D., Lance D., Stewart J. (2004) Medfly (Diptera: Tephritidae) genetic sexing: large scale field comparison of males-only and bisexual sterile fly releases in Guatemala, J. Econ. Entomol. 97, 1547–1553.
- Reuters (2008) http://www.reuters.com/article/environment News/idUSTRE49N38O20081024.
- Robinson A.S. (2002) Mutations and their use in insect control, Mut. Res. 511, 113–132.
- Robinson A.S., Hendrichs J. (2005) Prospects for the future development and application of the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated

pest management, Springer, Dordrecht, The Netherlands, pp. 727–760.

- Rössler Y. (1979) The genetics of the Mediterranean fruit fly: a "white pupae" mutant, Ann. Entomol. Soc. Am. 72, 583–585.
- Runner G.A. (1916) Effect of röntgen rays on the tobacco, or cigarette beetle and the results of experiments with a new form of röntgen tube, J. Agric. Res. 6, 383–388.
- Sakurai H., Murakami Y., Kohama T., Teruya T. (2000) Sterilizing mechanism of gamma-radiation in the female of West Indian potato weevil, *Euscepes postfasciatus*, Res. Bull. Fac. Agric. Gifu University, Japan 65, 13–20.
- Seth R.K., Barik T.K., Chauhan S. (2009) Interactions of entomopathogenic nematodes *Steinernema glaseri* (Rhabditida: Steinernematidae) cultured in irradiated hosts, with F<sub>1</sub> sterility: towards management of a tropical pest *Spodoptera litura* (Fabr.) (Lepidoptera: Noctuidae), Bio. Sci. Tech. 19, 139–155.
- Simmons G., Carpenter J.E., Suckling D.M., Addison M., Dyck V.A., Vreysen M.J.B. (2010) Improved quality management to enhance the efficacy of the sterile insect technique for lepidopteran pests. J. Appl. Entomol. 134, 261–273.
- Suckling D.M., Barrington A.M., Chhagan A., Stephens A.E.A., Burnip G.M., Charles J.G., Wee S.L. (2007) Eradication of the Australian painted apple moth *Teia anartoides* in New Zealand: trapping, inherited sterility and male competitiveness, in: Vreysen M.J.B., Robinson A.S., Hendrichs J. (Eds.), Area-wide control of insect pests. From research to field implementation, Springer, Dordrecht, The Netherlands, pp. 603–615.
- Staten R.L., Rosander R.W., Keaveny D.F. (1993) Genetic control of cotton insects: the pink bollworm as a working programme, in: Management of Insect Pests; Nuclear and Related Molecular and Genetic Techniques, Proceedings of a Symposium Organised by the FAO and IAEA, 19–23 October 1992, Vienna, Austria, IAEA, Vienna, Austria, pp. 269– 284.
- Toledo J., Rull J., Oropeza A., Hernández E., Liedo P. (2004) Irradiation of *Anastrepha obliqua* (Diptera: Tephritidae) revisited: optimizing sterility induction, J. Econ. Entomol. 97, 383–389.
- Tween G. (2002) MOSCAMED-Guatemala: an evolution of ideas, in: Barnes B.N. (Ed.), Proceedings of the 6th International Symposium on Fruit Flies of Economic Importance, 6–10 May 2002, Stellenbosch, South Africa, Isteg Scientific Publications, Irene, South Africa, pp. 119–126.
- United Press International (2008) http://www.upi.com/Science\_ News/2008/12/19/EU\_to\_ban\_most\_aerial\_crop\_spraying/ UPI-94951229708407/.
- US Working Group on the Food Crisis (2009) http://www. usfoodcrisisgroup.org/.
- Vale G.A., Torr S.J. (2005) User-friendly models of the costs and efficacy of tsetse control: application to sterilizing and insecticidal techniques, Med. Vet. Entomol. 19, 293–305.
- Van Borstel R.C. (1962) Effects of radiation on germ cells of insects: dominant lethals, gamete inactivation and gonial-cell killing, in: Radiation and Radioisotopes Applied to Insects of Agricultural Importance, Proceedings of a Symposium organised by FAO/IAEA, Athens, IAEA, Vienna, Austria, pp. 367–384.

- Van der Vloedt A.M.V., Taher M., Tenabe S.O. (1978) Effect of gamma radiation on the tsetse fly *Glossina palpalis palpalis* (Rob. Desv.) (Diptera: Glossinidae) with observations on the reproductive biology, Int. J. Appl. Radiat. Isot. 29, 713–716.
- van Driesche R., Hoddle M., Center T. (2008) Control of pests and weeds by natural enemies, Wiley-Blackwell, USA.
- van Lenteren J.C., Bale J., Bigler F., Hokkanen H.M.T., Loomans A.J.M. (2006) Assessing risks of releasing exotic biological control agents of arthropod pests, Annu. Rev. Entomol. 51, 609–634.
- Vargas-Terán M., Hofmann H.C., Tweddle N.E. (2005) Impact of screwworm eradication programmes using the sterile insect technique, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 629–650.
- Vreysen M.J.B. (1995) Radiation induced sterility to control tsetse flies. The effect of ionising radiation and hybridisation on tsetse biology and the use of the sterile insect technique in integrated tsetse control, Ph.D. dissertation, Wageningen Agricultural University, Wageningen, The Netherlands.
- Vreysen M.J.B. (2005) Monitoring sterile and wild insects in areawide integrated pest management programmes, in: Dyck V.A., Hendrichs J., Robinson A.S. (Eds.), Sterile insect technique. Principles and practice in area-wide integrated pest management, Springer, Dordrecht, The Netherlands, pp. 325–361.
- Vreysen M.J.B., Van der Vloedt A.M.V. (1995) Radiation sterilisation of *Glossina tachinoides* Westw. pupae: 1. The effect of dose fractionation and nitrogen during irradiation in the mid-pupal phase, Rev. Elev. Méd. Vét. Pays Trop. 48, 45–51.
- Vreysen M.J.B., Barclay H.J., Hendrichs J. (2006a) Modeling of preferential mating in area-wide control programs that integrate the release of strains of sterile males-only or both sexes, Ann. Entomol. Soc. Am. 99, 607–616.
- Vreysen M.J.B., Hendrichs J., Enkerlin W. (2006b) The sterile insect technique as a component of sustainable area-wide integrated pest management of selected horticultural insect pests, J. Fruit Ornam. Plant Res. 14, 107–131.
- Vreysen M.J.B., Gerardo-Abaya J., Cayol J.P. (2007b) Lessons from area-wide integrated pest management (AW-IPM) programmes with an SIT component: an FAO/IAEA perspective, in: Vreysen M.J.B., Robinson A.S., Hendrichs J. (Eds.), Area-wide control of insect pests. From research to field implementation, Springer, Dordrecht, The Netherlands, pp. 723–744.

- Vreysen M.J.B., Robinson A.S., Hendrichs J. (2007a) Area-wide control of insect pests, From research to field implementation, Springer, Dordrecht, The Netherlands.
- Vreysen M.J.B., Saleh K.M., Ali M.Y., Abdullah M.A., Zhu Z-R., Juma K.G., Dyck V.A., Msangi A.R., Mkonyi P.A., Feldmann H.U. (2000) *Glossina austeni* (Diptera: Glossinidae) eradicated on the island of Unguja (Zanzibar), using the sterile insect technique, J. Econ. Entomol. 93, 123–135.
- Vreysen M.J.B., Van der Vloedt A.M.V., Barnor H. (1996) Comparative gamma radiation sensitivity of *Glossina tachinoides* Westw., *Glossina fuscipes fuscipes* Newst. and *Glossina brevip alpis* Newst. (Diptera, Glossinidae), Int. J. Radiat. Biol. 69, 67–74.
- Wajnberg E., Scott J.K., Quimby P. (2001) Evaluating indirect ecological effects of biological control, CABI Publishing, CAB International, Oxon, UK.
- Walters M.L., Staten R.T., Roberson R.C. (2000) Pink bollworm integrated management using sterile insects under field trial conditions, Imperial Valley, California, in: Tan K.H. (Ed.), Area-Wide Control of Fruit Flies and Other Insect Pests, Proceedings of an International Conference on Area-Wide Control of Insect Pests, and the 5th International Symposium on Fruit Flies of Economic Importance, 28 May–5 June 1998, Penang, Malaysia, Penerbit Universiti Sains Malaysia, Pulau Pinang, Malaysia, pp. 201–206.
- Wong T.T.Y., Ramadan M.M., Herr J.C., McInnis D.O. (1992) Suppression of the Mediterranean fruit fly population with concurrent parasitoid and sterile fly releases in Kula, Maui, Hawaii, J. Econ. Entomol. 85, 1671–1681.
- Wyss J. (2000) Screwworm eradication in the Americas an overview, in: Tan K.H. (Ed.), Area-Wide Control of Fruit Flies and Other Insect Pests, Proceedings of an International Conference on Area-Wide Control of Insect Pests, and the 5th International Symposium on Fruit Flies of Economic Importance, 28 May–5 June 1998, Penang, Malaysia, Penerbit Universiti Sains Malaysia, Pulau Pinang, Malaysia, pp. 79–86.
- Zapater M.C., Andiarena C.E., Pérez Camargo G., Bartoloni N. (2009) Use of irradiated *Musca domestica* pupae to optimize mass rearing and commercial shipment of the parasitoid *Spalangia endius* (Hymenoptera: Pteromalidae), Bio. Sci. Tech. 19, S1, 261–270.
- Zeddies J., Schaab R.P., Neuenschwander P., Herren H.R. (2001) Economics of biological control of cassava mealybug in Africa, Agric. Econ. 24, 209–219.

# **Biodiversity and Pest Management in Orchard Systems**

Sylvaine Simon, Jean-Charles Bouvier, Jean-François Debras, and Benoît Sauphanor

### Contents

1	Introduction					
2	Orchard Systems and Biodiversity		695			
	2.1	Perennial Habitats	695			
	2.2	Multi-Strata Habitats	696			
	2.3	Plant Diversity in the Boundaries of Orchards	697			
	2.4	Pesticide Applications	697			
3	Benefits of Biodiversity for the Control of Orcha					
	Pests .		699			
	3.1	Manipulation of Plant Diversity to Enhance				
		Orchard Pest Control	699			
	3.2	Effect of Surrounding Land Uses	703			
	3.3	Predation by Insectivorous Birds	703			
4	Concl	usion	704			
<b>References</b>						

S. Simon (🖂)

e-mail: sylvaine.simon@avignon.inra.fr

Abstract Conventional agriculture is based on a high level of chemical inputs such as pesticides and fertilisers, leading to serious environmental impacts, health risks and loss of biodiversity in agrosystems. The reduction of pesticide use is a priority for intensively sprayed agricultural systems such as orchards. The preservation and promotion of biodiversity within orchards and their boundaries is therefore an issue to explore. Indeed, orchard systems contain high plant diversity and perennial multi-strata designs that provide wealthy resources and habitats to living communities such as beneficial organisms. Orchards thus offer favourable areas to maintain food-webs within the agrosystem, provided that favourable situations are not altered by cultural practices such as applying an excess of pesticides. Here, we analysed literature on the effects of the manipulation of plant diversity and habitats on the control of pests by arthropod and bird communities in apple, pear and peach orchards. Many investigations focus on the role of plant management to enhance biodiversity in orchards but only 22 research reports presenting 30 case studies were dedicated to the study of the ecosystem service provided by plant diversity for orchard pest control. The underlying mechanisms were seldom demonstrated, and the tested grass covers and tree assemblages aimed at favouring either the beneficial complex or only some beneficial species to control one or a few pests. The effect of plant management on pest control was mostly positive (16 cases) or null (9), but also negative in some cases (5). This finding reveals the difficulties of identifying selected plants or plant assemblages for the control of key pests. We conclude that further research is needed to identify the processes involved on different scales for biological control. Orchard systems

E. Lichtfouse et al. (eds.), Sustainable Agriculture Volume 2, DOI 10.1007/978-94-007-0394-0\_30,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Simon et al., Agron. Sustain. Dev. 30 (2010) 139–152. DOI: 10.1051/agro:2009013

INRA, UERI, Gotheron, 26320 Saint-Marcel-lès-Valence, France

should be re-designed to optimise ecosystem services provided by biodiversity.

**Keywords** Biodiversity • Orchard • Fruit tree • Plant • Arthropod • Bird • Community • Pest management • Hedgerow • Plant cover

#### 1 Introduction

Since the 90s and the Rio summit on biodiversity in 1992, there has been increasing concern about the environment, and a consciousness of the impact of production and service human activities on the environment and on biodiversity. Conventional agricultural production which is based on a high level of chemical inputs, e.g. pesticides and fertilisers, is in the focus. Political actions at national and European Community levels aim at reducing the number and amount of pesticides used (PIRRP, 2006; Commission Européenne, 2008), and to promote research programmes to reduce their use or the risks due to their use (Aubertot et al., 2005; Sauphanor et al., 2009). Indeed, many environmental risks are related to the use of conventional insecticides, e.g. their aerial dissemination and the contamination of soil and water, with negative effects on animal communities directly or indirectly exposed to these chemicals (Aubertot et al., 2005), and on human health (Baldi et al., 1998). Moreover, besides the loss of habitats, pesticides also contribute to the decrease in plant and animal biodiversity in the agrosystem (Krebs et al., 1999). Reduction in the use of plant protection products is thus crucial for the implementation of sustainable agricultural systems, and especially in systems based on a high pesticide use such as orchards. Indeed, in temperate areas, orchards are among the most intensively sprayed agricultural systems to impair pest and disease damage and produce fruits with no visible fault to satisfy international commercial quality standards. Whereas French orchards only represent 1% of the utilised agricultural area, they make up 21% of the insecticide sales in France (Codron et al., 2003). Recently, information on the pesticide residues in fruits has altered the perception of fruits by consumers as fresh and healthy food, leading in several countries to the implementation of zero residue programmes

S. Simon et al.

(Berrie and Cross, 2006). There is thus a challenge in satisfying a societal demand for environmentally friendly systems and healthy fruits, and keeping pests and diseases below economic thresholds to maintain the growers' income in an evolving regulation context. The preservation and promotion of biodiversity within agricultural landscapes could be a key issue to answer both ecological and agronomic purposes.

Cultural systems have, on different scales, a dualistic relationship with biodiversity. They often reduce, or alter, biodiversity through simplified systems or cultural practices. However, cultural systems also contribute to the agrosystem richness and to the occurrence of some plant and animal species that would otherwise have disappeared (Le Roux et al., 2008). Besides, cultural systems are dependent on several ecosystem processes provided by biodiversity that contribute to soil fertility, pollination and pest control (Zhang et al., 2007). If there is a consensus on the role of ecosystem services for crop production, then strategies to maintain, favour and preserve biodiversity are more debated. These strategies can range from surface areas dedicated to biodiversity conservation ('land-sparing agriculture') to biodiversity preservation within agricultural areas ('eco-friendly agriculture') (Clergue et al., 2005). Agroecology (Altieri, 1995) represents the challenge to match production and biodiversity conservation within agricultural landscapes, especially in areas where cultivated lands occupy a large surface area.

Among cultivated crops, orchards are particularly suitable systems to study the level of ecosystem services provided by biodiversity, because they are perennial systems and present a complex multi-strata design. In most European orchards, plant components associated with fruit trees are planted and/or preserved and managed within orchards and/or in their boundaries for agronomic purposes, e.g. prevention of soil compaction due to machinery traffic, windbreaks and physical barriers. Orchard plant design thus contributes to plant diversity within agricultural areas and therefore to an increase in resources for animal communities such as arthropods and birds (Boller et al., 2004), among which are pest antagonists, provided that cultural practices, namely pesticide use, are not disruptive. Moreover, the importance of the ecosystem service provided by natural enemies for pest control has been pointed out for decades in orchards, with

a focus on the control of many fruit pests such as mites, aphids, leafminers and psyllids by natural enemies (e.g. Wildbolz, 1988; Boller et al., 2004). The key role of natural enemies has been demonstrated for psyllids in pear orchards (Shaltiel and Coll, 2004) and mites in apple orchards (Solomon et al., 2000). There is thus a need to evaluate in orchards the role of functional diversity, i.e. the ecosystem service for pest control, but also the importance of disservices due to agricultural and ecosystemic management (Zhang et al., 2007).

The aim of the present review is to analyse the complex relationships between orchard systems, i.e. orchard design and practices, and functional biodiversity with a focus on plant, arthropod and bird communities. Our work is based on a thorough investigation of the ISI Web of Knowledge database from 1992 to January 2008, and on former articles cited in this literature, complemented by recently accepted articles. Orchards were considered to be agricultural surface areas dedicated to fruit production, which excluded several types of agroforestry or pastoral systems planted with fruit trees. Only pome and stone fruit productions in temperate areas were studied. Lastly, biodiversity was defined according to Noss (1990) and comprised compositional, structural and functional biodiversity on different scales. We develop two main points: (i) the contribution of orchard systems to plant and animal diversity against adverse effects of orchard practices on biodiversity, and (ii) the benefits for the orchard pest management of biological control through the conservation of habitats.

#### 2 Orchard Systems and Biodiversity

The potential contribution of orchard systems to biodiversity is based on the analysis of their main specific features: permanency of the system, multi-strata design and adjacent plant management (Table 1) as favourable aspects, and the need for intensive pest management, including a recurrent use of pesticides, as a detrimental factor.

### 2.1 Perennial Habitats

Orchards are planted for several years or even decades in temperate areas: the diversity in time due to the successive crops of the cultural rotation is thus low in orchard systems. However, such a permanency in the host plant and the associated cultural practices is likely to enhance the stability of the system (Brown and Welker, 1992) and its resilience (Kozár, 1992). The permanency of the host plant favours the presence of some herbivores (which include pests) to the benefit of the permanency of food-webs. The entomofauna richness measured in apple orchards is higher than in annual crops in Hungary (Kozár, 1992), and the control provided by the natural enemies of pests is also reported to be higher in perennial than in annual crops (Hall and Ehler, 1979; Risch et al., 1983). Food and living conditions in organic apple orchards are favourable to the presence and the nesting of insectivore birds such as the Great Tit Parus major (Paridae),

Plant diversity provided by the agrosystem	Annual systems	Standard orchard systems
Diversity in time	Crop rotation: turnover of host plants	Perennial crop: permanency of host plants
Diversity in space:		
- Plant architecture	-	Fruit tree complex branching structures
- Spatial distribution of plants within the field	Homogeneous plant cover and mono-stratum systems	Heterogeneous distribution (fruit tree rows, alleys) and multi-strata (arboreal, understorey) systems
- Adjacent diversity	Large fields adapted to machinery	Need for windbreaks in windy regions: planting of lining hedgerows

 Table 1
 Contribution of annual and orchard systems to plant biodiversity

with reproduction rates equal to that of pesticide-free forests (Bouvier et al., 2005). Lastly, the soil litter is also likely to develop, to the benefit of the scavenger biomass which favours the abundance of some natural enemies (Longcore, 2003). As permanent habitats, orchards therefore contribute to the presence of a diversified arthropod community including scavengers, herbivores, predators and parasitoids, and to the permanency of food-webs including high trophic levels, namely, the insectivorous bird community.

#### 2.2 Multi-Strata Habitats

In most cases both understorey and arboreal habitats are present within orchards. A grassy ground cover is generally sown or naturally occurs between rows and in the turning ends of the orchard to prevent soil compaction by machinery traffic, to limit erosion and/or pesticide transfer (Lacas et al., 2005). The arboreal habitat mainly consists of planted fruit trees. Various species of arthropods can live in one or more of these strata. The surface area of the orchard thus consists of a meadow interplanted with fruit tree rows where the soil arthropod community is more related to that of a meadow than of a forest (Fazekas et al., 1992). The effect of the orchard plant design on arthropod diversity is analysed on different spatial scales: (i) arboreal strata and within-tree structure, (ii) additional grass cover as understorey strata, and (iii) orchard system.

Fruit trees have a characteristic architecture, partly due to tree-training performed to ensure regular fruit bearing. Indeed, branching structures of the branches and patterns of distribution and growth of fruiting and vegetative shoots contribute to a complex withintree architecture. Moreover, different scales, including leaf structure and infra-structures such as domatia and trichomes (Cortesero et al., 2000) are present. This structural complexity favours the richness of the entomocenosis (Price et al., 1980; Lawton, 1983) and the abundance of natural enemies (Langellotto and Denno, 2004; Finke and Denno, 2006). Even though underlying processes affecting tree arthropods within complex structures are not always disentangled by authors the most plausible are: (i) the diversity of plant resources benefits specialised herbivores which are the prey or the host of various natural enemies, being themselves the prey or host of other predatory or parasitoid insects; and (ii) intra-guild predation decreases in complex structures (Finke and Denno, 2006) and 'enemy-free spaces' are more important (Lawton, 1983). The structural complexity and heterogeneity of the fruit tree thus favours the diversity of the canopy arthropod community. However, complex structures are also detrimental to the foraging of some beneficial insects, through an increase in the time needed to locate their prey or host (Gingras and Boivin, 2002; Skirvin, 2004). Consequently, pest control through parasitism or predation is not always higher in complex than in simpler structures despite higher entomological diversity or abundance (Langellotto and Denno, 2004; Simon et al., 2007a).

As additional plant strata to productive trees, the plant cover in the alleys is generally composed of grasses (Poaceae), mixed with weeds and sometimes with other sown species such as leguminous plants. The presence of an understorey cover generally provides a benefit for the orchard pest control. A threespecies plant cover sown in the alleys provides a higher richness and diversity of the pear canopy entomocenosis compared with a bare ground (Rieux et al., 1999). The beneficial aphidiphagous complex of the apple orchard is favoured by flowering strips to the benefit of aphid control (Wyss, 1995). Plant covers are only reported to be detrimental compared with a bare ground in peach orchards where they favour leafhoppers (McClure et al., 1982) and phytophagous mites (Meagher and Meyer, 1990a).

The co-existence of different strata creates a diversity of habitats and resources for animals: shelters, e.g. resting, diapause or hibernating sites, reproduction areas and refuge areas to escape disruptive agricultural practices, as well as food, e.g. alternate preys and hosts, nectar and pollen (Greaves and Marshall, 1987). Arthropod communities exploiting the soil, the grass and the canopy (Miliczky et al., 2000) cohabit within the orchard and contribute to its richness. Some species are likely to exploit more than one of these strata and are therefore likely to stay and multiply through higher levels of resources. As high levels of beneficial arthropod richness are displayed within the grass cover, whereas low levels of predation and pest control are observed within the arboreal strata (Simon et al., 2007b), strong interactions among strata are not always established in field experiments. The hypothesis of a structural rather than a functional assemblage is promoted by some authors (Vogt et al., 1998; Miliczky et al., 2000; Horton et al., 2002; Simon et al., 2007b).

It cannot be excluded that the beneficial complex of the fruit tree canopy does not benefit the grass cover richness or diversity. Although the intrinsic complexity of fruit trees and the diversity in resources provided by the orchard plant design are high, the resulting arthropod diversity is not always highly functional for pest control.

# 2.3 Plant Diversity in the Boundaries of Orchards

As fruits are delicate high value products and orchards perennial systems, they easily support the installation of windbreaks in windy regions (Prokopy, 1994). The most common ones are planted hedgerows. Some of the planted hedgerows in the orchard boundaries are multi-species hedgerows, for instance, composed after the recommendations by the IDF (Institut pour le Développement Forestier, 1981). As the diversity of planted fruit species or cultivars is low within the orchards, e.g. the most common cases are one or a few clones, these hedgerows improve the orchard system plant diversity. Although hedgerows may impair crop protection by harbouring potential pests and diseases (Solomon, 1981; Jeanneret, 2000), they are also physical barriers that stop drifts from adjacent pesticide applications and thus minimise side effects of pesticide use. If the hedgerow contributes by itself to the local plant biodiversity through the same mechanisms as plant covers, the association of orchards and hedgerows within the landscape creates and favours specific habitats and ecosystems. This contributes to the increase in global and landscape biodiversity (Pollard and Holland, 2006). A mosaic landscape consisting of orchards separated by hedgerows and/or ditches favours a specific flora and fauna through a higher availability of habitats and resources (Rands, 1986). Such areas are also hibernating sites for many insect species (Harwood et al., 1992; Lys and Nentwig, 1994). The communities of both adjacent plantings and local surroundings interfere with the orchard (Krebs et al., 1999; Simon, 1999; Benton et al., 2003). On a local scale, the biodiversity of the orchard system is improved by lining hedgerows and ditches (Green et al., 1994; Parish et al., 1994; Moles and Breen, 1995), as the biodiversity on the landscape scale is improved (Benton et al., 2003) through an increase

in available biotopes (Rosenzweig, 1995). This latter aspect is especially emphasised for mobile taxa such as Lepidoptera (Jonsen and Fahrig, 1997) and birds (Robertson et al., 1990). Orchard systems and their boundaries are thus highly relevant candidates to contribute plant and animal diversity on different scales.

However, it is necessary to minimise such potentially favourable situations: the widespread use of the mating disruption method to control Lepidoptera requires large surface areas without interplanted hedgerows (Witzgall et al., 2008; Sauphanor, 2009). Italian and Northern American studies (Neumann, 1993) indicate an optimal efficiency for continuous surface areas of homogeneous orchards of tens of hectares protected by this method, which favours pheromone diffusion and minimises the vulnerability of borders. A recent study (Ricci et al., 2009) also indicates that the codling moth Cydia pomonella populations of a given orchard are negatively correlated with the surrounding surface areas planted with apple and treated with chemicals, which promote the production of apples within large surface areas whatever the pest control method (if efficient) against codling moth. The planting of large surface area orchards excluding hedgerows, which are replaced by hail nets as windbreaks and shields for the physical control of C. *pomonella* and tortricids (Tasin et al., 2008), are likely to develop in Southern France. There is an antagonism between the optimal use of various pest control methods and the management of plant diversity in the boundaries of the orchard system.

#### 2.4 Pesticide Applications

Because of their host-tree permanency pests and diseases may remain present in the orchard throughout the year. This favours the increase in infestation or infection levels from one year to another, with the need for a continuous protection, namely, a recurrent use of pesticides to control them. Fruit tree protection is highly intensive and requires far more pesticide amounts than other crops. In 2006, an average of 36.5 treatments were sprayed in French apple orchards (Sauphanor et al., 2009). In all producing countries, current apple production systems resort to such intensive use of pesticides (Eurostat, 2002). Moreover, the trend is for an increase in the yearly number of treatments because of the development of resistant strains in some pests (Sauphanor et al., 2000; Reyes et al., 2008), low surface areas planted with resistant or low-susceptibility cultivars, and 'zero default fruit' market standards. Global warming is also likely to increase voltinism and the period of risks for some pests (Sauphanor, 2004), and to introduce new pests. From green tip to harvest, i.e. during a 6- to 8-month period, apple orchards are thus under pest and disease management regimes based on the use of pesticides. The side effects of their use on organisms living or foraging within the orchard may be direct through mortality and/or lower fecundity, or indirect through biomass (i.e. prey or host) reduction or host-plant suppression in the food-web.

In orchards the effect of pesticides and pest management regimes on arthropods is well documented for a few taxonomic groups, amongst which are spiders (Pekár, 1999; Bogya and Markó, 1999; Bogya et al., 1999; Miliczky et al., 2000; Brown et al., 2003) and ground-living beetles (Pearsall and Walde, 1995; Labrie et al., 2003) but it is more seldom studied for the total arthropod community (Sauphanor et al., 1993, 2005; Suckling et al., 1999; Brown and Schmitt, 2001; Debras et al., 2006; Simon et al., 2007b). The use of pesticides has a negative effect on hunting spiders (Pekár, 1999), ground-living arthropods (Epstein et al., 2000) and insects parasitising leaf miners (Prokopy et al., 1996) but, surprisingly, the total arthropod diversity or richness of the tree canopy is not or very little affected by the use of broad-spectrum insecticide programmes compared with more environmentally friendly methods (Suckling et al., 1999; Brown and Schmitt, 2001; Simon et al., 2007b). Hypotheses that are likely to explain such results may be related to: the resilience of the orchard system (Brown, 1993); a high immigrating rate of arthropods in small-sized orchards within mosaic landscapes (Liss et al., 1986; Whalon and Croft, 1986; Brown, 1993; Kozár, 1992; Bengtsson et al., 2005; Miliczky and Horton, 2005); and/or the inadequacy of synthetic diversity indices to give information on a whole community composed of groups with inconsistent responses (Suckling et al., 1999; Hole et al., 2005). However, even though the diversity measured by classical ecological indices such as the Shannon index is not always affected, the abundance of arthropods is always negatively affected by intensive pest management regimes (Suckling et al., 1999; Brown and Schmitt, 2001; Simon et al., 2007b), as are soil micro-arthropods (Doles et al., 2001).

The structure of the arthropod community also differs among high- and low-intensity pest management regimes (Andreev et al., 2006), and the natural control of some apple pests may be altered under intensive management regimes (Brown and Adler, 1989; Balázs et al., 1996; Suckling et al., 1999; Simon et al., 2007b): the beneficial arthropod complex is thus no longer present for ecosystem services. Enhanced ecosystem services for pest control permitted by a reduction in pesticide exposition such as in organic or low-input orchards illustrate the mutual benefits between conservation biological control and a reduced pesticide use (Sauphanor and Audemard, 1983; Brown, 2001a; Zehnder et al., 2007).

Birds constitute bio-indicators which are used to assess the effect of cultural practices on the environment (Ormerod and Watkinson, 2000). Because they occupy a high or top position in the food-web, they are relevant indicators of its global alterations (Furness and Greenwood, 1993). Besides, some of their biological requirements such as reproduction are concomitant with the period of pesticide applications in fields (Chamberlain et al., 2000). The documentation on the effect of pest management strategies on birds in apple orchards is still incomplete. Most of the studies focus on the reproduction rate of passerine birds, which is lower in intensively managed orchards compared with organic ones, in Northern America (Powell, 1984; Fluetsch and Sparling, 1994; Bishop et al., 2000) as well as in Europe (Bouvier et al., 2005). The effect of pesticides on bird communities is less studied. The bird diversity and abundance in German orchards was higher in organic than in Integrated Pest Management (IPM) orchards (Rösler, 2003). Consistently, the two research teams working on the subject have assessed that bird communities are more abundant and diversified in organic apple orchards, or to a lesser extent in IPM orchards, than in conventional intensive ones where the number of insectivore species is also lower (Bouvier, 2004; Genghini et al., 2006).

Local and regional environments, cultural practices and manipulations of the orchard plant diversity widely interfere. Only a few recent studies (Debras et al., 2006; Agerberg, 2007; Monteiro et al., 2008) have quantified the weight of these external factors. The weight of environmental variables to explain the composition of the orchard arthropod community is 28.7%,

whereas it is 12.4% and 2.2% for cultural practices and lining hedgerows, respectively (Debras et al., 2006). For bird communities, both pesticide applications and the local environment account for 25% in the results, whereas the landscape effect contributes 15% to the total variance in apple orchards in Southern France (Agerberg, 2007). This latter result is consistent with the study by Monteiro et al. (2008) on the parasitoid community of pome fruit orchards in Southern France that displayed a similar contribution of local (27%) and landscape (16%) factors. However, further research is needed to validate such results within various contexts and regions, to assess the potential contribution of local and/or landscape diversity to explain the structure of bird and arthropod communities, and to identify local and landscape managements maximising the abundance of natural enemies.

# 3 Benefits of Biodiversity for the Control of Orchard Pests

The most studied benefits of biodiversity for fruit tree production are related to crop protection and are mainly based on an increase in plant diversity that favours the increase in animal diversity, including birds, mammals and arthropods. A higher level of pest control is thus expected, at least for some pests, through an increase in the abundance and the richness of their natural enemies. Within this framework of conservation biological control (Barbosa, 1999), we develop the effect on orchard pest control of (i) two plant assemblages associated with the orchard, i.e. plant ground covers and lining hedgerows, and (ii) the local land uses in the agricultural landscape. Lastly, the role of insectivore birds, favoured by nesting-boxes, will be discussed. Diversity is understood as measurements by classical ecological indices such as the Shannon index, but also by richness and abundance of the studied groups. All taxonomic levels are taken into account.

# 3.1 Manipulation of Plant Diversity to Enhance Orchard Pest Control

The effect of plant diversity on the arthropod populations of pests and natural enemies relies on several complex mechanisms (Russell, 1989): plant-insect relationships, prey-predator and host-parasitoid interactions, population dynamics, and structure and organisation of arthropod communities (Liss et al., 1986). The mechanisms involved are seldom demonstrated by authors. Both bottom-up and top-down effects are promoted to explain a reduced herbivory in complex compared with simple environments (Russell, 1989). The following mechanisms are either simultaneously (Bugg and Waddington, 1994) or individually proposed:

- within a diversified system, the decrease in pest damage is related to greater difficulties in localising their host plant(s) and to lower resources (Risch et al., 1983); structural and chemical complexities of plant assemblages are thus the cause of such decrease in herbivory (Brown, 1998);
- plant associations may alter the microclimate, the physiological stage or even the pest biology, contributing to pest control (Parfait and Jarry, 1987; Andow, 1991);
- due to a diversified vegetation, the predatory and parasitoid complex likely to control pests is maintained and made perennial (Risch et al., 1983; Szentkiràlyi and Kozár, 1991; Chaubet, 1993; Wyss, 1996; Brown, 2001a). The longevity or fecundity of some species may also be increased (Irvin et al., 2006).

The manipulation of the orchard plant diversity may affect communities living within or near the orchard through an increase in the resource range, i.e. habitat, shelter and food. Herbivores, including orchard pests, polyphagous and disease vector arthropods, pollinators, and predatory and parasitoid arthropods are involved, and the manipulation can result in beneficial or detrimental effects for the orchard pest control (Grison and Biliotti, 1953; Van Emden and Williams, 1974; Gruys, 1982; Fye, 1983; Solomon, 1981; Bugg and Waddington, 1994; Prokopy, 1994; Rieux, 1994; Schoemans, 1995; Simon, 1999; Boller et al., 2004; Debras, 2007). Very few studies address the economic benefit of such manipulation of plant diversity in the orchard or its boundaries. Besides several studies on the arthropod community of understorey plants (e.g. Westigard et al., 1990; Flexner et al., 1991; Meyer et al., 1992; Coli et al., 1994 on mites), pest control resulting from the introduction of plant assemblages is seldom directly assessed. The results may

Fruit tree		Plant manipulation(s)	Effect on pest	
production	Target pest(s)	or presence	control <sup>1</sup>	Source/Region
Apple	Apple aphids Aphis pomi	Flower strips	Positive	Wyss (1995); Wyss et al. (1995),
	Dysaphis plantaginea		Positive	Switzerland
Apple	Apple aphids	Flower strips	Positive	Pfammater and Vuignier (1998),
	D. plantaginea			Switzerland
Apple	Apple aphids A. pomi	Flower strips	Null	Vogt et al. (1998); Vogt and Weigel
	D. plantaginea	Negative		( <b>1999</b> ), Germany
Apple	Tent caterpillar and	Understorey plants	Positive	Leius (1967), USA
	codling moth		Positive	
Apple	Leafroller (Tortricidae)	Buckwheat	Positive	Stephens et al. (1998), New Zealand
Apple	Leafroller (Tortricidae)	Buckwheat	Positive	Irvin et al. (2006), New Zealand
		Alyssum	Positive	
		Phacelia	Null	
Apple	Leafroller (Tortricidae)	Peach nectaries	Null	Brown et al. (2008), USA
Apple	Apple aphids	Peach nectaries	Negative	Spellman et al. (2006), USA
	Aphis spiraecola	Buckwheat	Null	(greenhouse experiment)
Apple	Apple pests	Plant cover and/or	Null or variable	Brown (2001b), USA
Peach	Peach pests	interplanted fruit-tree	according to	
			years and pests	
Apple	Apple pests	Plant cover	Null	Jenser et al. (1999)
Apple	Spider mites	Understorey plants	Positive under conditions	Croft B.A. (1982), USA
Apple	Spider mite	Understorey plants	Positive under	Alston D. (1994), USA
11	Tetranychus spp.		conditions	
Apple	Spider mite	Plant cover	Null	Nyrop et al. (1994), USA
11	Panonychus ulmi			
Apple	Apple pests	Plant cover	Globally	Altieri and Schmidt (1985), USA
			positive	
Apple	Spider mite P. ulmi	Adjacent bushes	Positive	Tuovinen (1994), Finland
Apple	Spider mites	Plant cover	Positive	Yan et al. (1997), China
	Tetranychus spp.			
Apple	Spider mite P. ulmi	Flower plant mixture	Null	Fitzgerald and Solomon (2004), UK
Pear	Cacopsylla pyri		Null	
Peach	leafhoppers	Plant cover	Negative	McClure et al. (1982), USA
Peach	Spider mite T. urticae	Plant cover	Negative	Meagher and Meyer (1990a), USA
Peach	Hemiptera species	Plant cover	Negative	Meagher and Meyer (1990b), USA
Pear	C. pyri	Plant cover	Positive	Rieux et al. (1999), France
Pear	C. pyri	Hedgerow	Positive	Debras (2001, 2007), France

**Table 2** Effects of plant diversity on the control of orchard pests

<sup>1</sup> The effect of plant manipulation on pest control is considered to be positive, null or negative when either the density of the pest arthropod of the fruit tree, fruit damage and/or the number of pesticide applications against the target pest is lower, equal or higher, respectively, compared with control.

vary according to the host fruit species, the pest and the tested plant assemblage (Table 2). Among the 22 listed articles presenting 30 case studies on the subject, the effect on pest control was positive in 16 cases, 5 plant assemblages had a negative effect and 9 others were indifferent. Plant manipulations generally aimed at favouring either predator or parasitoid beneficial taxonomic groups or species. The total beneficial complex is more seldom targeted. Most of the plant manipulations were based on the manipulation of understorey plants or plant assemblages, or on the analysis of naturally occurring plant ground covers. Only two of them were related to arboreal plant assemblages (adjacent bushes or lining hedgerows), attesting to the difficulties of carrying out field experiments on perennial plant assemblages. With the exception of one case in orchards, i.e. the detrimental effect of flower strips on apple aphid (Vogt and Weigel, 1999), negative effects were mainly due to the development in weeds of spider mites migrating into fruit trees when weeds are chemically or mechanically removed.

Several aspects can explain such variability in the results: the studied "pest-antagonist" couple, local context, composition, age and management of the tested plant assemblage, and orchard design. Cultivar and age of producing orchards are reported to be of little importance to explain the structure of the orchard arthropod community, providing the cultivars are not insect-resistant and tree architecture is similar (Brown and Adler, 1989). Only juvenile orchards, which are seldom experimented on, differ from older ones (Pekár, 2003). More generally, the effect of a plant manipulation largely relies on the biology of each targeted pest and each natural enemy, and on their interactions, between them and with other species of the arthropod community. We present below understorey and arboreal plant manipulations dedicated to enhancing pest control in orchards. Only plant-based approaches were considered. The manipulation of the habitat of ground-dwelling arthropods, for instance, by mulching the groundcover (Miñarro and Dapena, 2003; Mathews et al., 2004) was not reviewed.

#### 3.1.1 Lining Hedgerows

Hedgerows lining the orchard are plant assemblages comprising tree species that may constitute a reservoir, or a source, of natural enemies, and also the source of infestation or infection by pests and diseases (Solomon, 1981; Prokopy, 1994; Schoemans, 1995; Maudsley, 2000; Boller et al., 2004). Studies or reviews on the specific entomocenosis of many tree species planted in hedgerows are available in Southern France (Barthelet, 1982; Defrance et al., 1987; Campo, 1992; Carraretto, 1992; Gauthier, 1993; Simon et al., 1993; Rieux, 1994; Delmas, 1995; Sarthou, 1995; Reboulet, 1996; Simon, 1999; Baudry et al., 2000; Debras, 2001; Debras et al., 2002), but precise and comprehensive information is still missing because of local specificity, climatic variations and timeconsuming assessments by experts trained in arthropod systematics.

Very few hedgerows dedicated to crop protection have been experimented on. The mixed hedgerow proposed by Rieux (1994) for the control of the pear psyllid *Cacopsylla pyri* in pear orchards and experimented on since 1992 in Southern France has been built up according to the following principles and experimentally assessed (Simon et al., 2009):

- exclude tree species hosting orchard or quarantine pests and diseases, i.e. hawthorn, which is the host of fireblight;
- provide some natural enemies, i.e. the one(s) active against the main orchard pest(s) with various habitats and resources: shelter, hibernating site, and areas to escape within-crop cultural practices. These consist of hollow stems of herbaceous plants, bark crevices, evergreen leaves of bush or tree species, intertwine stems of creeping species; food such as pollen, nectar, alternate preys or hosts;
- organise all year long successive resources in order to maintain and multiply beneficial arthropods in the vicinity of the orchard;
- favour the motion of natural enemies from the hedgerow towards the orchard, using tree species hosting migrating alternate preys which induce natural enemies to search for new preys.

The presence of natural enemies is generally higher in the part of the orchard lining the hedgerow than in its centre (Altieri and Schmidt, 1986; Reboulet, 1996; Paoletti et al., 1998) and aphid abundance is correlatively the lowest in orchard edges where beneficial numbers are the highest (Altieri and Schmidt, 1986). A gradient of density from the hedgerow towards the orchard is described for lacewings (Rodet, 1985; Simon et al., 1998). Earwigs issuing from the hedgerow are collected within the orchard in Southern France (Debras et al., 2007). Debras (2007) and Debras et al. (2008) also assessed that the distribution of natural enemies within the orchard is affected by the hedgerow: natural enemies actively move from the hedgerow to the orchard in relation to prey availability, even though cultural practices (among which the use of pesticides) alter this functional pattern in most orchards. However, a significant effect of the hedgerow on the orchard beneficial complex is not always displayed. Coli et al. (1994) did not relate high densities of predatory mites hosted by bush species to mite populations of the adjacent orchard. As an adverse effect, codling moth abundance was observed to be the highest along the hedgerow (Audemard, 1992). Lastly, no significant correlation was displayed between predator abundance due to plant environment and C. pyri control in a survey of 8 commercial pear orchards (Simon, 1999). Patterns of pest distribution and patterns of predator densities introduced by the plant environment are thus not always correlated. Such discrepancies may be explained by the thermal and biological requirements of the considered arthropod species and by the climatic effect of the hedgerow on the distribution of both pests and natural enemies (Debras et al., 2008; Ricci et al., 2009). Lastly, the age (Burgio et al., 2006) and the cultural management of these hedgerows may alter their structural and plant diversity, therefore altering their functionality (Forman and Baudry, 1984). The benefit of the increase in the abundance and diversity of natural enemies induced by plant manipulation is, however, seldom measured; the benefit, if any, can be slight, and not sufficient to avoid pesticides against the most noxious pests. We noticed that only pests such as mites and psyllids, which can be tolerated at high population levels in the orchards, may benefit from such manipulations (Table 2). Beyond shortterm pest control, the recolonisation of the orchard by natural enemies issuing from adjacent plant assemblages can contribute to the restoration of the community structure (e.g. the case of mites, see Tuovinen, 1994) and to a more stable system.

#### 3.1.2 Plant Ground Covers and Interplanted Fruit Tree Species

A wide range of plant covers and interplanted peach trees (Brown, 2001b; Brown et al., 2008) were tested by the authors (Table 2). Most of the tested assemblages aimed at providing beneficial arthropods with pollen and nectar through flowers or peach nectaries. Grassy or flower strips sown in the orchard alleys (between rows) are proposed in apple orchards to help control the rosy apple aphid Dysaphis plantaginea (Wyss et al., 1995; Wyss, 1995; Pfammater and Vuignier, 1998; Vogt et al., 1998). Single species covers with buckwheat, phacelia or alyssum were experimented on in New Zealand to help control Tortricidae (Stephens et al., 1998; Irvin et al., 2006). Weeds are also companion plants in orchards and may shelter natural enemies (Kozár et al., 1994), especially flower weeds (Leius, 1967; Zandstra and Motooka, 1978; Wyss, 1995) and nettle Urtica dioica (Stary, 1983; Hérard, 1986). The plant cover not only shelters an abundant arthropod community likely to offer alternate preys or hosts, but also orchard

pests: aphids, mites (Meagher and Meyer, 1990a), phytophagous mirids (Fye, 1980), leafhoppers (McClure et al., 1982; Meagher and Meyer, 1990b), tortricids (Brown, 2001b) and Coleoptera (Wyss, 1996). However, the migration of these pests towards the cultivated trees is often more affected by the management of the plant cover than by the plant cover itself: weeding leads the hosted pests to migrate towards another resource, i.e. orchard trees (Van Emden and Williams, 1974; McClure et al., 1982; Westigard et al., 1990; Flexner et al., 1991). Natural enemies hosted by the plant cover of the alleys are also negatively affected by frequent mechanical mowing (Horton et al., 2002).

For the total arthropod fauna and beneficial complex, the presence of a grassy ground cover within the orchard increases (Altieri and Schmidt, 1985) or not (Wyss, 1996) the diversity of beneficial arthropods. Responses in terms of pest control vary widely according to the pests (Altieri and Schmidt, 1985; Brown and Glenn, 1999), and most of the studied plant covers address one or a few key pests. Many studies in Northern America focus on the effect of understorey covers on mite populations in orchards. Information is provided on the conditions for optimal biological control of spider mites by predatory mites in terms of surface area to be covered by grass, distance to fruit trees, and composition of the plant assemblage (Nyrop et al., 1994; Croft B.A., 1982; Alston D., 1994). In pear orchards, both an increase in Anthocorid numbers and a decrease in C. pyri prey are assessed when a grassy ground cover is sown in the alleys compared with bare ground (Rieux et al., 1999). Flower strips in apple orchards enhance D. plantaginea control (Wyss et al., 1995; Pfammater and Vuignier, 1998). By providing a within-orchard higher density of preys they contribute to maintaining a high density of spiders and generalist predators, which prey on immigrating aphids in autumn, and can survive on, and control, low densities of preys. In spring, flower strips are also expected to favour aphidiphagous syrphids (Wyss, 1995), but this effect is not displayed in a second experiment (Vogt et al., 1998; Vogt and Weigel, 1999), most probably because of a delayed bloom in a Northern region and the mowing of the strips in winter. Lastly, very few studies (Irvin et al., 2006) address all the processes involved in the tri-trophic system targeted by plant manipulations: food preference of each orchard pest, effect of companion plants on the

abundance and the fitness of the natural enemies to promote, predation or parasitism rates in the orchard and interactions with other natural enemies. The plant ground cover is therefore a plant component of the orchard which is easily manipulated and experimented on. A wide range of plant assemblages targeting various pests has already been tested with promising results. As for hedgerows, we notice that empirical, rather than scientific, knowledge is involved in most cases. Consequently, beyond adverse effects due to the management of the plant cover, failure or success cases in pest control cannot be explained and results are not always reproducible. Further research is needed to identify occurring processes and the ability of both pests and beneficial arthropod species to exploit both understorey and arboreal resources.

#### 3.2 Effect of Surrounding Land Uses

The association of both agricultural and uncultivated areas has been considered to preserve biodiversity (Grison and Biliotti, 1953) and to favour natural enemies of crop pests (Chaubet, 1993; Sarthou, 1995; Landis et al., 2000; Deguine and Ferron, 2004; Tscharntke et al., 2007). Integrated Fruit Protection (OILB, 1977) and ecological compensation areas (Garnier, 1994; Herzog et al., 2005) rely on such association in order to enhance natural control of orchard pests and to increase biodiversity, respectively. The effect of local or regional landscape on the arthropod populations of orchards is reported by many authors (Liss et al., 1986; Altieri and Schmidt, 1986; Whalon and Croft, 1986; Szentkirályi and Kozár, 1991; Kozár, 1992; Bengtsson et al., 2005; Miliczky and Horton, 2005). Winged arthropods represent 50% of the total number of the orchard entomofauna and are strongly related to local and regional backgrounds (Szentkiràlyi and Kozár, 1991; Kozár, 1992). However, very few studies address the relationships or correlation between the features of the agricultural area and beneficial effects or pest control in orchards. The total surface area covered by uncultivated lands (woods, fallow fields) within 100 m around pear orchards is significantly and positively correlated with the total and beneficial arthropod diversity of the orchard, and negatively correlated with C. pyri pest numbers, whereas the types of crops and their relative importance within the same area are not significantly correlated. Hedgerows lining the orchards increase the above-mentioned correlation between surface areas of uncultivated lands and the orchard beneficial complex diversity (Simon, 1999). These results are consistent with the study by Gut et al. (1988), establishing that the development of *C. pyri* is low in plant diversified environments. Such approaches are similar to those based on landscape ecology and developed in vineyards (Van Helden et al., 2006).

The presence of alternate host plants of pests in the orchard surroundings permits the provision of refuge areas for insecticide-susceptible alleles. This is likely to contribute to the management of resistance to insecticides and therefore to the sustainability of crop protection methods, provided that susceptible alleles have a selective advantage on resistant ones when insecticides are no longer applied. For the rosy apple aphid whose primary host is the apple tree, as well as the codling moth hosted by cultivated fruit trees only (apple, pear, nut and quince trees), refuge zones for strains susceptible to insecticides are mainly unmanaged or organic orchards (Boivin et al., 2005). More than plant diversity, it is the diversity of cultural practices which is expected to be beneficial. Lastly, it is clear that the study and the management scales outstrip the orchard scale, due to the potential moving distances of various groups of arthropods (Lewis, 1969).

#### 3.3 Predation by Insectivorous Birds

The integration of insectivorous birds into pest control patterns is of benefit with higher yields and income (Jones et al., 2005a). Such integration can be performed by increasing the number of artificial nesting sites within the orchard, especially for cavity-nesting passerine birds which lack natural cavities (Bishop et al., 2000). Sanz (2001) showed that such management of artificial nesting sites favours the installation of Tit populations. As all species of Tits feed their brood with Lepidoptera caterpillars, they may significantly reduce fruit damage caused by codling moth larvae (Mols et al., 2005). As birds generally avoid parasitised preys, the biological control of pests due to birds seems to be complementary to that of other bio-control agents (Jones et al., 2005b).

Beyond a global increase in the richness of the agrosystem, an increase in predation or parasitism favoured by the conservation of habitats of beneficial organisms is assessed in many studies. There are generally only partial effects for pest control which is insufficient to reduce the use of pesticides except for some pests, e.g. mites and psyllids, that can be tolerated at high levels of populations without any damage on fruits or reduction in yield. Further research is needed to investigate all of the processes involved in conservation biological control on different interconnected scales and to identify: (i) the most relevant beneficial candidates or association of candidates to be promoted among predators and parasitoids, generalists and specialists; and (ii) the species composition, age, density and design of plant assemblages that would maximise beneficial effects and minimise detrimental ones when considering the global orchard community. The reduction in pesticide exposition of orchard communities is certainly a key point to maximise ecosystem services for pest control. We also promote the redesign of orchard systems to meet such a purpose of an 'agroecologic' orchard. We particularly propose to investigate: (i) the effect of a decrease in the genetic (one clone) and spatial (linear arrangements) monotony of current orchard designs; (ii) the emphasis of some favourable traits of current orchard designs, i.e. multi-strata design, plant diversified environment and soil litter development. Options such as the interplanting of missing strata (i.e. bush layer), mix cropping coupled with the supervised management of the whole orchard plant diversity, and the management of the soil organic status are candidate issues to favour functional diversity for pest control; and (iii) the manipulation of the architectural and microclimatic traits of the fruit tree through genetics and tree training, as a tool to modify the habitat of orchard pests and the foraging area of their natural enemies, and therefore their development. Lastly, the challenge mainly relies on integrating all these tools on different interconnected scales, from fruit tree leaf infrastructures to orchard and landscape scales, in order to maximise ecosystem services on each scale and to implement synergistic effects.

Acknowledgements The authors wish to thank C. Bussi for his advice on an earlier draft of this manuscript. We thank Katie L. and David Bennett for improving the English of the manuscript.

#### References

- Agerberg J. (2007) Impact des pratiques agricoles et du paysage sur l'avifaune des vergers de pommiers du Sud-Est de la France, Mémoire INRA Avignon – AgroParisTec, 57 p. + annexes.
- Alston D. (1994) Effect of apple orchard floor vegetation on density and dispersal of phytophagous and predaceous mites in Utah, Agr. Ecosyst. Environ. 50, 73–84.
- Altieri M.A. (1995) Agroecology: the science of sustainable agriculture, Westview Process, Boulder, USA
- Altieri M.A., Schmidt L.L. (1985) Cover crop manipulation in Northern California orchards and vineyards: effects on arthropods communities, Biol. Agric. Hortic. 3, 1–24.
- Altieri M.A., Schmidt L.L. (1986) The dynamics of colonizing arthropod communities at the interface of abandoned, organic and commercial apple orchards and adjacent woodland habitats, Agr. Ecosyst. Environ. 16, 29–43.
- Andow D.A. (1991) Polycultures and pest populations, Annu. Rev. Entomol. 36, 561–586.
- Andreev R., Olszak R., Kutinkova H. (2006) Harmful and beneficial entomofauna in apple orchards grown under different management systems, Bull. IOBC/wprs 29, 13–19.
- Aubertot J.N., Barbier J.M., Carpentier A., Gril J.J., Guichard L., Lucas P., Savary S., Savini I., Voltz M. (2005) Pesticides, agriculture et environnement, Réduire l'utilisation des pesticides et limiter leurs impacts environnementaux, Expertise scientifique collective, synthèse du rapport, INRA & Cemagref, France, 64 p.
- Audemard H. (1992) Population dynamics in codling moth, in: Van der Geest L.P.S., Evenhuis H.H. (Eds.), Tortricid pests: their biology, natural enemies and control, Elsevier Science Publishers, Amsterdam, pp. 329–338.
- Balázs K., Jenser G., Bujáki G. (1996) Eight years' experiences of IPM in Hungarian apple orchards, Bull. IOBC/wprs 19, 95–101.
- Baldi I., Mohammed-Brahim B., Brochard P., Dartigues J.-F., Salamon R. (1998) Long-term effects of pesticides on health: review of current epidemiologic knowledge, Rev. Epidemiol. Sante 46, 134–142.
- Barbosa P. (1999) Conservation Biological Control, Academic Press, San Diego, USA, 396 p.
- Barthelet B. (1982) Étude faunistique d'une haie brise-vent composite, Mémoire de l'École Nationale des Ingénieurs des Travaux Horticoles, Angers, 101 p.
- Baudry O., Bourgery C., Guyot G., Rieux R. (2000) Les haies composites réservoirs d'auxiliaires, Ctifl, Paris, 116 p.
- Bengtsson J., Ahnström J., Weibull A.C. (2005) The effects of organic agriculture on biodiversity and abundance: a metaanalysis, J. Appl. Ecol. 42, 261–269.
- Benton T.G., Vickery J.A., Wilson J.D. (2003) Farmland biodiversity: is habitat heterogeneity the key? Trends Ecol. Evol. 18, 182–188.
- Berrie A., Cross J. (2006) Development of an integrated pest and disease management system for apples to produce fruit free from pesticide residues - aspects of disease control, Bull. IOBC/wprs 29, 129–138.
- Bishop C.A., Collins B., Mineau P., Burgess N.M., Read W.F., Risley C. (2000) Reproduction of cavity-nesting birds in pesticide-sprayed apple orchards in southern Ontario, Canada, 1988–1994, Environ. Toxicol. Chem. 19, 588–599.

- Bogya S., Markó V. (1999) Effect of pest management systems on ground-dwelling spider assemblages in an apple orchard in Hungary, Agr. Ecosyst. Environ. 73, 7–18.
- Bogya S., Markó V., Szinetár Cs. (1999) Comparison of pome fruit orchard inhabiting spider assemblages at different geographical scales, Agr. Forest Entomol. 1, 261–269.
- Boivin T., Chadoeuf J., Bouvier J.-C., Beslay D., Sauphanor B. (2005) Modelling the interactions between phenology and insecticide resistance genes in the codling moth *Cydia pomonella*, Pest Manag. Sci. 61, 53–67.
- Boller E.F., Häni F., Poehling H.M. (2004) Ecological infrastructures: ideabook on functional biodiversity at the farm level, Landwirtschaftliche Beratungszentrale Lindau, Lindau, Suisse.
- Bouvier J.C. (2004) Exposition et réponse de l'avifaune aux pratiques phytosanitaires en vergers : Possibilités de traduction en termes de bioindication et de transfert vers la profession agricole, Mémoire d'Ingénieur Diplômé par l'État, spécialité Agriculture, ENSA Montpellier, 62 p. + annexes.
- Bouvier J.C., Toubon J.F., Boivin T., Sauphanor B. (2005) Effects of apple orchard management strategies on the great tit (*Parus major*) in Southeastern France, Environ. Toxicol. Chem. 24, 2846–2852.
- Brown M.W. (1993) Resilience of the natural arthropod community on apple to external disturbance, Ecol. Entomol. 18, 169–183.
- Brown M.W. (1998) Diversification of orchard ecosystems to augment populations of biological control fauna, in: Brunnhofer V., Soldan T. (Eds.), Proc. 6th European Congress of Entomology (Book of Abstracts), Ceské Budejovice, August 23–29 1998, Institute of Entomology, Academy of the Czech republic and University of South Bohemia, pp. 623–624.
- Brown M.W. (2001a) Functional biodiversity and agroecosystems management: 2. role in integrated fruit production, Bull. IOBC/wprs 24, 5–11.
- Brown M.W. (2001b) Flowering ground cover plants for pest management in peach and apple orchards, Bull. IOBC/wprs 24, 379–382.
- Brown M.W., Adler C.R.L. (1989) Community structure of phytophagous arthropods on apple, Environ. Entomol. 18, 600–607.
- Brown M.W., Welker W.V. (1992) Development of the phytophagous arthropod community on apple as affected by orchard management, Environ. Entomol. 31, 485–492.
- Brown M.W., Glenn D.M. (1999) Ground cover plants and selective insecticides as pest management tools in apple orchards, J. Econ. Entomol. 92, 899–905.
- Brown M.W., Schmitt J.J. (2001) Seasonal and diurnal dynamics of beneficial insect populations in apple orchards under different management intensity, Biol. Control 30, 415–424.
- Brown M.W., Schmitt J.J., Abraham B.J. (2003) Seasonal and diurnal dynamics of spiders (Araneae) in West Virginia orchards and the effect of orchard management on spider communities, Environ. Entomol. 32, 830–839.
- Brown M.W., Mathews C.R., Krawczyk G. (2008) Analyzing the results of biodiversity experiments: enhancing parasitism of tufted apple budmoth, in: Proc. 7th IOBC International Conference on Integrated Fruit Production (Book of Abstracts), Avignon, October 28–30 2008, p. 6.

- Bugg R.L., Waddington C. (1994) Using cover crops to manage arthropod pests of orchards: a review, Agr. Ecosyst. Environ. 50, 11–28.
- Burgio G., Ferrari R., Boriani L., Pozzati M., van Lenteren J. (2006) The role of ecological infrastructures on Coccinellidae (Coleopera) and other predators in weedy field margins within northern Italy agroecosystems, Bull. Insect. 59, 59–67.
- Campo L. (1992) La faune auxiliaire des haies composites : inventaire sur quatre sites du sud de la France, Mémoire de l'Institut Polytechnique, Toulouse, 64 p.
- Carraretto L. (1992) Les haies composites : intérêt dans la protection intégrée, Mémoire d'ingénieur de l'École Supérieure d'Agriculture de Purpan, 89 p.
- Chamberlain D.E., Fuller R.J., Bunce R.G.H., Duckworth J.C., Shrubb M. (2000) Patterns of change in the abundance of farmland birds in relation to the timing of recent intensification of agriculture in England and Wales, J. Appl. Ecol. 37, 771–788.
- Chaubet B. (1993) Diversité écologique, aménagement des agroécosystèmes et favorisation des ennemis des cultures : cas des aphidiphages, Courr. Cell. Environ. INRA 18, 45–63.
- Clergue B., Amiaud B., Pervanchon F., Lasserre-Joulin F., Plantureux S. (2005) Biodiversity: function and assessment in agricultural areas. A review, Agron. Sustain. Dev. 25, 1–15.
- Codron J.M., Jacquet F., Habib R., Sauphanor B. (2003) Bilan et perspectives environnementales de la filière arboriculture fruitière, in: Quae (Ed.), Les Dossiers de l'Environnement de l'INRA 23: Agriculture, territoire, environnement dans les politiques européennes, pp. 31–67.
- Coli W.M., Ciurlino R.A., Hosmer T. (1994) Effect of understory and border vegetation composition on phytophagous and predatory mites in Massachussetts commercial apple orchard, Agr. Ecosyst. Environ. 50, 49–60.
- Commission Européenne (2008) http://ec.europa.eu/food/ plant/protection/pesticides/legislation\_fr.htm consulté le 10/03/2009.
- Cortesero A.M., Stapel J.O., Lewis W.J. (2000) Understanding and manipulating plant attributes to enhance biological control, Biol. Control 17, 35–49.
- Croft B.A. (1982) Management of apple orchard weeds to improve biological control of spider mites, Abstracts Meet, Weed Sci. Soc. Am. 257, 134.
- Debras J.-F. (2001) Optimisation du choix des essences d'une haie composite pour lutter contre le psylle du poirier *Cacopsylla pyri* L., Mémoire d'Ingénieur Diplômé par l'État, ENSA Montpellier, 80 p. + annexes.
- Debras J.-F. (2007) Rôles fonctionnels des haies dans la régulation des ravageurs : Le cas du psylle *Cacopsylla pyri* L. dans les vergers du sud-est de la France, Thèse de doctorat de l'Université d'Avignon, Sciences de la vie, 239 p.
- Debras J.-F., Cousin M., Rieux R. (2002) Mesure de la ressemblance de la faune utile du poirier avec celle de 43 espèces végétales pour optimiser la composition de haies réservoirs d'auxiliaires, Fruits 57, 55–65.
- Debras J.-F., Torre F., Rieux R., Kreiter S., Garcin M.S., Van Helden M., Buisson E., Dutoit T. (2006) Discrimination between agricultural management and the hedge effect in pear orchards (south-eastern France), Ann. Appl. Biol. 149, 347–355.

- Debras J.-F., Dussaud A., Rieux R., Dutoit T. (2007) Recherche prospective sur le rôle "source" des haies en production fruitière intégrée. Le cas des perce-oreilles : *Forficula auricularia* L. et *Forficula pubescens* Gené, C.R. Acad. Sci. Fr. 330, 664–673.
- Debras J.-F., Senoussi R., Rieux R., Buisson E., Dutoit T. (2008) Spatial distribution of an arthropod community in a pear orchard (southern France), Identification of a hedge effect, Agr. Ecosyst. Environ. 127, 166–176.
- Defrance H., Marboutie G., Atger P. (1987) Expérimentation d'une haie brise-vent composite, PHM Rev. Hortic. 280, 25–29.
- Deguine J.-P., Ferron P. (2004) Protection des cultures et développement durable, bilan et perspectives, Courrier Cell. Env. INRA 52, 57–65.
- Delmas Y. (1995) Peuplements d'Arthropodes d'un verger de pêchers, des haies de bordure et de la strate herbacée de l'inter-rang, Mémoire de l'École Nationale des Ingénieurs des Travaux de l'Horticulture et du Paysage, Angers, 56 p.
- Doles J.L., Zimmerman R.J., Moore J.C. (2001) Soil microarthropod community structure and dynamics in organic and conventionally managed apple orchards in Western Colorado, USA, Appl. Soil Ecol. 18, 83–96.
- Epstein D.L., Zack R.S., Brunner J.F., Gut L., Brown J.J. (2000) Effects of broad-spectrum insecticides on epigeal arthropod biodiversity in Pacific Northwest apple orchards, Environ. Entomol. 29, 340–348.
- Eurostat (2002) The use of plant protection products in the European Union, Data 1992–1999, Eurostat report 2002.
- Fazekas J., Kadar F., Lovei G.L. (1992) Comparison of ground beetle assemblages (Coleoptera: Carabidae) of an abandoned apple orchard and the bordering forest, Acta Phytopathol. Entomol. Hung. 27, 233–238.
- Finke D.L., Denno R.F. (2006) Spatial refuge from intraguild predation: implications for prey suppression and trophic cascades, Oecologia 149, 265–275.
- Fitz Gerald J.D., Solomon M.G. (2004) Can flowering plants enhance numbers of beneficial arthropods in UK apple and pear orchards? Biocontrol Sci. Techn. 14, 291–300.
- Flexner J.L., Westigard P.H., Gonzalves P., Hilton R. (1991) The effect of groundcover and herbicide treatment on twospotted spider mite density and dispersal in southern Oregon pear orchards, Entomol. Exp. Appl. 60, 111–123.
- Fluetsch K.M., Sparling D.W. (1994) Avian nesting success and diversity in conventionally and organically Managed orchards, Environ. Toxicol. Chem. 13, 2118–2124.
- Forman R.T.T., Baudry J. (1984) Hedgerows and hedgerow networks in Landscape Ecology, Environ. Manag. 8, 495–510.
- Furness R.W., Greenwood J.J.D. (1993) Birds as monitors of environmental change, Chapman and Hall, London, 288 p.
- Fye R.E. (1980) Weed sources of *Lygus* bugs in the Yakima valley and Columbia basin in Washington, J. Econ. Entomol. 73, 469–473.
- Fye R.E. (1983) Cover crop manipulation for building pear psylla (Homoptera: Psyllidae) predator populations in pear orchards, J. Econ. Entomol. 76, 306–310.
- Garnier M. (1994) Milieux naturels servant à la compensation écologique, Document environnement n°17, OFEFP, Berne, 35 p.
- Gauthier J. (1993) Haies composites et strate herbacée: incidences sur les populations d'arthropodes d'un verger de

pêchers conduit en protection intégrée dans la moyenne vallée du Rhône, Mémoire de l'École Nationale des Ingénieurs des Travaux de l'Horticulture et du Paysage, Angers, 58 p.

- Genghini M., Gellini S., Gustin M. (2006) Organic and integrated agriculture: the effects on bird communities in orchard farms in northern Italy, Biodivers. Conserv. 15, 3077–3094.
- Gingras D., Boivin G. (2002) Effect of plant structure, host density and foraging duration on host finding by *Trichogramma evanescens* (Hymenoptera: Trichogrammatidae), Environ. Entomol. 31, 1153–1157.
- Greaves M.P., Marshall E.J.P. (1987) Field margins: definitions and statistics, in: Way J.M., Greig-Smith P.W. (Eds.), Field margins: definitions and statistics, British Crop Protection Council, Farnham, UK, pp. 3–10.
- Green R.E., Osborne P.E., Sears E.J. (1994) The distribution of passerine birds in hedgerows during the breeding season in relation to characteristics of the hedgerow and adjacent farmland, J. Appl. Ecol. 31, 677–692.
- Grison P., Biliotti E. (1953) La signification des "stationsrefuges" pour la faune entomologique, C.R. Acad. agric. Fr. 39, 106–109.
- Gruys P. (1982) Hits and misses. The ecological approach to pest control in orchards, Entomol. Exp. Appl. 31, 70–87.
- Gut L.J., Westigard P.H., Liss W.J. (1988) Arthropod colonization and community development on young pear trees in Southern Oregon, Melandaria 46, 1–13.
- Hall R.W., Ehler L.E. (1979) Rate of establishment of natural enemies in classical biological control, Bull. Entomol. Soc. Am. 25, 280–282.
- Harwood R.W.J., Wratten S.D., Nowakoski M. (1992) The effect of managed fields margin on hoverfly (Diptera: Syrphidae) distribution and within fields abundance, in: Brighton Crop Protection Conference – Pests and Diseases. British Crop Protection Council, Hampshire, UK, pp. 1033–1037.
- Hérard F. (1986) Annotated list of the entomophagous complex associated with pear psylla, *Psylla pyri* (L.) (Hom.: Psyllidae) in France, Agronomie 6, 1–34.
- Herzog F., Dreier S., Hofer G., Marfurt C., Schupbach B., Spiess M., Walter T. (2005) Effect of ecological compensation areas on floristic and breeding bird diversity in Swiss agricultural landscapes, Agr. Ecosyst. Environ. 108, 189–204.
- Hole D.G., Perkins A.J., Wilson J.D., Alexander I.H., Grice P.V., Evans A.D. (2005) Does organic farming benefit biodiversity? Biol. Conserv. 122, 113–130.
- Horton D.R., Broers D.A., Lewis R.R., Granatstein D., Richard S.Z., Unruh T.R., Moldenke A.R., Brown J.J. (2002) Effects of mowing frequency on densities of natural enemies in three Pacific Northwest pear orchards, Entomol. Exp. Appl. 106, 135–145.
- IDF Institut pour le Développement Forestier (1981) Réalisation pratique des haies brise-vent et bandes boisées, IDF, Paris, 129 p.
- Irvin N.A., Scarratt S.L., Wratten S.D., Frampton C.M., Chapman R.B., Tylianakis J.M. (2006) The effects of floral understoreys on parasitism of leafrollers (Lepidoptera: Tortricidae) on apples in New Zealand, Agr. Forest Entomol. 8, 25–34.
- Jeanneret P. (2000) Interchanges of a common pest guild between orchards and surrounding ecoystems: a multivariate analysis of landscape influence, in: Ekbom B., Irwin M.E., Robert Y. (Eds.), Interchanges of insects between agricultural

and surrounding landscapes, Kluwer Academic, Dordrecht, Netherlands, pp. 85–107.

- Jenser G., Balázs K., Erdélyi Cs., Haltrich A., Kádár F., Kozár F., Markó V., Rácz V., Samu F. (1999) Changes in arthropod population composition in IPM apple orchards under continental climatic conditions in Hungary, Agr. Ecosyst. Environ. 73, 141–154.
- Jones G.A., Sieving K.E., Jacobson S.K. (2005a) Avian diversity and functional insectivory on North-Central Florida farmlands, Conserv. Biol. 19, 1234–1245.
- Jones G.A., Sieving K.E., Jacobson S.K., Avery M.L., Meagher R.L. (2005b) Parasitized and non-parasitized prey selectivity by an insectivorous bird, Crop Prot. 24, 185–189.
- Jonsen I.D., Fahrig L. (1997) Responses of generalist and specialist insect herbivores to landscape spatial structure, Landscape Ecol. 12, 185–197.
- Kozár F. (1992) Organization of arthropod communities in agroecosystems, Acta Phytopathol. Entomol. Hung. 27, 365–373.
- Kozár F., Brown M.W., Lightner G. (1994) Spatial distribution of homopteran pests and beneficial insects in an orchard and its connection with ecological plant protection, J. Appl. Entomol. 117, 519–529.
- Krebs J.R., Wilson J.D., Bradbury R.B., Siriwardena G.M. (1999) The second silent spring? Nature 400, 611–612.
- Labrie G., Prince C., Bergeron J.M. (2003) Abundance and developmental stability of *Pterostichus melanarius* (Coleoptera: Carabidae) in organic and integrated pest management orchards of Quebec, Canada, Environ. Entomol. 32, 123–132.
- Lacas J.-G., Voltz M., Gouy V., Carluer N., Gril J.-J. (2005) Using grassed strips to limit pesticide transfer to surface water: a review, Agron. Sustain. Dev. 25, 253–266.
- Landis D.A., Wratten S.D., Gurr G.M. (2000) Habitat management to conserve natural enemies of arthropod pests in agriculture, Annu. Rev. Entomol. 45, 175–201.
- Langellotto G.A., Denno R.F. (2004) Responses of invertebrate natural enemies to complex-structured habitats: a metaanalytical synthesis, Oecologia 139, 1–10.
- Lawton J.H. (1983) Plant architecture and the diversity of phytophagous insects, Ann. Rev. Entomol. 28, 23–39.
- Leius K. (1967) Influence of wild flowers on parasitism of tent caterpillar and codling moth, Can. Entomol. 99, 444–446.
- Le Roux X., Barbault R., Baudry J., Burel F., Doussan I., Garnier E., Herzog F., Lavorel S., Lifran R., Roger-Estrade J., Sarthou J.-P., Trommetter M. (2008) Agriculture et biodiversité. Valoriser les synergies. Expertise scientifique collective, synthèse du rapport, INRA (France).
- Lewis T. (1969) The diversity of the insect fauna in a hedgerow and neighbouring fields, J. Appl. Ecol. 6, 453–458.
- Liss W.J., Gut L.J., Westigard P.H., Warren C.E. (1986) Perspectives on arthropod community structure, organization, and development in agricultural crops, Annu. Rev. Entomol. 31, 455–478.
- Longcore T. (2003) Terrestrial arthropods as indicators of ecological restoration success in coastal sage scrub (California, USA), Restor. Ecol. 11, 397–409.
- Lys J.A., Nentwig W. (1994) Improvement of the overwintering sites for Carabidae, Staphylinidae and Araneae by stripmanagement in a cereal field, Pedobiologia 38, 238–242.

- Mathews C.R., Bottrell D.G., Brown M.W. (2004) Habitat manipulation of the apple orchard floor to increase grounddwelling predators and predation of *Cydia pomonella* (L.) (Lepidoptera: Tortricidae), Biol. Control 30, 265–273.
- Maudsley M. J. (2000) A review of the ecology and conservation of hedgerow invertebrates in Britain, J. Environ. Manage. 60, 65–76.
- McClure M.S., Andreadis T.G., Lacy G.H. (1982) Manipulating orchard ground cover to reduce invasion by leafhopper vectors of peach X-disease, J. Econ. Entomol. 75, 64–68.
- Meagher R.L., Meyer J.R. (1990a) Influence of ground cover and herbicide treatments on *Tetranychus urticae* populations in peach orchards, Exp. Appl. Acarol. 9, 149–158.
- Meagher R.L., Meyer J.R. (1990b) Effect of ground cover management on certain abiotic and biotic interactions in peach orchard ecosystems, Crop Prot. 9, 65–72.
- Meyer J.R., Zehr E.I., Meagher R.L. Jr., Salvo S.K. (1992) Survival and growth of peach trees and pest populations in orchard plots managed with experimental ground covers, Agr. Ecosyst. Environ. 41, 353–363.
- Miliczky E.R., Horton D.R. (2005) Densities of beneficial arthropods within pear and apple orchards affected by distance from adjacent native habitat and association of natural enemies with extra-orchard host plants, Biol. Control 33, 249–259.
- Miliczky E.R., Calkins C.O., Horton D.R. (2000) Spider abundance and diversity in apple orchards under three insect pest management programmes in Washington State, USA, Agr. Forest Entomol. 2, 203–215.
- Miñarro A., Dapena T. (2003) Effects of groundcover management on ground beetles (Coleoptera: Carabidae) in an apple orchard, Appl. Soil Ecol. 23, 111–117.
- Moles R.T., Breen J. (1995) Long-term change within lowland farmland bird communities in relation to field boundary attributes, Biology Environment - Proc. R. Ir. Acad. 95, 203–215.
- Mols C.M.M., van Noordwijk A.J., Visser M.E. (2005) Assessing the reduction of caterpillar numbers by great tits *Parus major* breeding in apple orchards, ARDEA 93, 259–269.
- Monteiro L.B., Dor C., Franck P., Lavigne C., Sauphanor B. (2008) Pest management practices and environmental factors affect natural regulation of the codling moth, in: Proc. 7th IOBC International Conference on Integrated Fruit Production (Book of Abstracts), Avignon, October 28–30, 2008, p. 99.
- Neumann U. (1993) How to achieve better results with the mating disruption technique, Bull. IOBC/wprs 16, 93–98.
- Noss R.F. (1990) Indicators for monitoring biodiversity: a hierarchical approach, Conserv. Biol. 4, 355–364.
- Nyrop J.P., Minns J.C., Herring C.P. (1994) Influence of ground cover on dynamics of *Amblyseius fallacis* Garman (Acarina: Phytoseiidae) in New York apple orchards, Agr. Ecosyst. Environ. 50, 61–72.
- OILB Organisation Internationale de Lutte Biologique (1977) Vers la production agricole intégrée par la lutte intégrée, Bull. IOBC/wprs 4, 1–163.
- Ormerod S.J., Watkinson A.R. (2000) Editors' Introduction: Birds and agriculture, J. Appl. Ecol. 37, 699–705.
- Paoletti M.G., Boscolo P., Sommaggio D. (1998) Beneficial insects in fields surrounded by hedgerows in North Eastern Italy, Biol. Agric. Hortic. 15, 311–323.

- Parfait G., Jarry M. (1987) Diversité végétale et impact des insectes phytophages : une revue bibliographique des méthodes appliquées au cas des cultures associées, Acta Oecol., Oecol. Gen. 8, 365–378.
- Parish T., Lakhani K.H., Sparks T.H. (1994) Modelling the relationship between bird population variables and hedgerow and the other field margin attributes: species richness of winter, summer and breeding birds, J. Appl. Ecol. 31, 764–775.
- Pearsall I.A., Walde S.J. (1995) A Comparison of Epigaeic Coleoptera Assemblages in Organic, Conventional, and Abandoned Orchards in Nova-Scotia, Canada, Can. Entomol. 127, 641–658.
- Pekár S. (1999) Effect of IPM practices and conventional spraying on spider population dynamics in an apple orchard, Agr. Ecosyst. Environ. 73, 155–166.
- Pekár S. (2003) Change in the community of epigeal spiders and harvestmen (Araneae, Opiliones) with the age of an apple orchard, Plant Soil Environ. 49, 81–88.
- Pfammatter W., Vuignier R. (1998) Amélioration de la lutte biologique dans les cultures fruitières au moyen de bandes de plantes sauvages, in : 1<sup>er</sup> Colloque transnational sur les luttes biologique, intégrée et raisonnée, Lille, January 21–23, 1998, Région Nord-Pas-de-Calais, Lille, France, pp. 71–72.
- PIRRP (2006) Plan Interministériel de Réduction des Risques liés aux Pesticides 2006-2009, Ministère de l'Environnement, de l'Énergie, du Développement durable et de l'Aménagement du territoire, http://www.ecologie. gouv.fr/Plan-interministeriel-de-reduction.html, consulté le 10/03/2009.
- Pollard K.A., Holland J.M. (2006) Arthropods within the woody element of hedgerows and their distribution pattern, Agr. Forest Entomol. 8, 203–211.
- Powell G.V.N. (1984) Reproduction of an atricial songbird, the redwinged blackbird, in field treated with organophosphate insecticide fenthion, J. Appl. Ecol. 21, 83–95.
- Price P.W., Bouton C.E., Gross P., McPheron B.A., Thompson J.N., Weis A.E. (1980) Interactions among three trophic levels: influence of plants on interactions between insect herbivores and natural enemies, Ann. Rev. Ecol. Syst. 11, 41–65.
- Prokopy R.J. (1994) Integration in orchard pest and habitat management, Agr. Ecosyst. Environ. 50, 1–10.
- Prokopy R.J., Mason J.L., Christie M., Wright S.E. (1996) Arthropod pest and natural enemy abundance under secondlevel versus first-level integrated pest management practices in apple orchards: a 4-year study, Agr. Ecosyst. Environ. 57, 35–47.
- Rands M.R.W. (1986) Effect of hedgerow characteristics on partridge breeding densities, J. Appl. Ecol. 23, 479–487.
- Reboulet J.N. (1996) Bandes boisées ou enherbées : incidence sur les ravageurs et les auxiliaires des cultures voisines, Adalia 33, 22–24.
- Reyes M., Franck P., Olivares J., Margaritopoulos J., Knight A., Sauphanor B. (2008) Worldwide variability of insecticide resistance mechanisms in the codling moth, *Cydia pomonella* L. (Lepidoptera: Tortricidae), Bull. Entomol. Res., doi:10.1017/S0007485308006366.
- Ricci B., Franck P., Toubon J.-F., Bouvier J.-C., Sauphanor B., Lavigne C. (2009) The influence of landscape on insect pest dynamics: a case study in southeastern France, Landscape Ecol. 24, 337–349.

- Rieux R. (1994) Et si l'on pouvait aménager l'environnement végétal des cultures pour manipuler les auxiliaires? Fruit Belge 447, 9–16.
- Rieux R., Simon S., Defrance H. (1999) Role of hedgerows and ground cover management on arthropod populations in pear orchards, Agr. Ecosyst. Environ. 73, 129–140.
- Risch S.J., Andow D., Altieri M.A. (1983) Agroecosystem diversity and pest control: data, tentative conclusions, and new research directions, Environ. Entomol. 12, 625–629.
- Robertson G.M., Eknert B., Ihse M. (1990) Habitat analysis from infrared aerial photographs and the conservation of birds in Swedish agricultural landscape, Am. Bio. 19, 195–203.
- Rodet G. (1985) Incidence des zones refuge sur la dynamique de l'entomofaune d'un verger de pêchers, Mémoire de DEA, Université de Paris XI, Paris VI et Museum, 66 p.
- Rosenzweig M.L. (1995) Species diversity in space and time, Cambridge University Press, Cambridge, UK.
- Rösler S. (2003) Natur- und Sozialverträglichkeit des Integrierten Obstbaus, Ph.D. thesis University of Kassel, Germany, 430 p.
- Russell E.P. (1989) Enemies hypothesis: a review of the effect of vegetational diversity on predatory insects and parasitoids, Environ. Entomol. 18, 590–599.
- Sanz J.J. (2001) Experimentally increased insectivorous bird density results in a reduction of caterpillar density and leaf damage to Pyrenean oak, Ecol. Res. 16, 387–394.
- Sarthou J.P. (1995) Haies composites et protection biologique: l'entomofaune associée aux essences ligneuses, in: Corroyer N., Garapon D., Arnal A. (Eds.), Mise en place et développement de haies composites en arboriculture biologique, Groupe de Recherche en Agriculture Biologique, Avignon, France, Annexe 2.
- Sauphanor B. (2004) Réchauffement climatique et nuisibilité du carpocapse sur les pommiers, in: Lésel R. (Ed.), Effets du réchauffement climatique, Cahiers d'études et de recherches francophones / Agricultures 13.
- Sauphanor B. (2009) Les phéromones d'insectes et la lutte par confusion sexuelle, in: Pintureau B., Grenier S., Mouret H., Sauge M.H., Sauphanor B., Sforza R., Tailliez P., Volkoff A.N. (Eds.), La lutte biologique, application aux arthropodes ravageurs et aux adventices, Ed. Ellipse, Coll. Technosup, Paris, pp. 52–73.
- Sauphanor B., Audemard H. (1983) Analyse comparée des populations de Lépidoptères en en vergers de Pomacées par piégeage avec des phéromones de synthèse, Agronomie 3, 947–955.
- Sauphanor B., Miniggio C., Faivre d'Arcier F. (1993) Effets à moyen terme des pesticides sur la faune auxiliaire en vergers de poiriers, J. Appl. Entomol. 116, 467–478.
- Sauphanor B., Brosse V., Bouvier J.-C., Speich P., Micou A., Martinet C. (2000) Monitoring resistance to diflubenzuron and deltamethrin in French codling moth populations (*Cydia pomonella*), Pest. Manag. Sci. 56, 74–82.
- Sauphanor B., Bouvier J.C., Boisneau C., Rieux R., Simon S., Capowiez Y., Toubon J.F. (2005) Impacts biologiques des systèmes de protection en vergers de pommiers, Phytoma Def. Veg. 581, 32–36.
- Sauphanor B., Dirwimmer C. et al. (2009) Analyse comparative de différents systèmes en arboriculture fruitière, in: Ecophyto R&D : vers des systèmes de culture économes

en produits phytosanitaires, Rapport d'Expertise Collective Inra, Inra Ed., Tome IV, 49 p.

- Schoemans P. (1995) Intérêt des insectes et araignées présents sur des haies vis-à-vis de vergers de pommiers conduits en lutte intégrée, Fruit Belge 456, 117–123.
- Shaltiel L., Coll M. (2004) Reduction of pear psylla damage by the predatory bug *Anthocoris nemoralis* (Heteroptera: Anthocoridae): the importance of orchard colonization time and neighbouring vegetation, Biocontrol Sci. Techn. 14, 811–821.
- Simon S. (1999) Incidence de l'environnement végétal sur les populations d'arthropodes du verger de poiriers, Thèse de doctorat de l'Université de Montpellier 2, Biologie des populations et écologie, 438 p.
- Simon S., Rieux R., Faivre d'Arcier F., Defrance H., Comte D. (1993) Aménagement de l'environnement végétal du verger de poiriers, in: ANPP Association Nationale de Protection des Plantes, 3<sup>e</sup> Conférence Internationale sur les Ravageurs en Agriculture, Montpellier, December 7–9 1993, ANPP, Paris, pp. 1009–1016.
- Simon S., Defrance H., Rieux R., Reboulet J.N. (1998) Les bandes boisées, réservoirs d'arthropodes : incidence sur la protection des cultures, Gibier Faune Sauvage, Game Wildl. 15(HS1), 33–42.
- Simon S., Sauphanor B., Lauri P.-E. (2007a) Control of fruit tree pests through manipulation of tree architecture, Pest Techn. 1, 33–37.
- Simon S., Defrance H., Sauphanor B. (2007b) Effect of codling moth management on orchard arthropods, Agr. Ecosyst. Environ. 122, 340–348.
- Simon S., Sauphanor B., Defrance H., Lauri P.E. (2009) Manipulation des habitats du verger biologique et de son environnement pour le contrôle des bio-agresseurs. Des éléments pour la modulation des relations arbre-ravageur-auxiliaires, Innov. Agron. 4, 125–134.
- Skirvin D.J. (2004) Virtual plant models of predatory mite movement in complex plant canopies, Ecol. Model. 171, 301–313.
- Solomon G. (1981) Windbreaks as a source of orchard pests and predators, in: Thresh J.M. (Ed.), Pests, pathogens and vegetation: the role of weeds and wild plants in the ecology of crop pests and diseases, Pitman Books Ltd., London, pp. 273–283.
- Solomon M.G., Cross J.V., Fitz Gerald J.D., Campbell C.A.M., Jolly R.L., Olszak R.W., Niemczyk E., Vogt H. (2000) Biocontrol of pests of apples and pears in northern and central Europe - 3. Predators, Biocontrol Sci. Techn. 10, 91–128.
- Spellman B., Brown M.W., Mathews C.R. (2006) Effect of floral and extrafloral resources on predation of *Aphis spiraecola* by *Harmonia axyridis* on apple, BioControl 51, 715–724.
- Stary P. (1983) The perennial stinging nettle (*Urtica dioica*) as a reservoir of aphid parasitoïds (Hymenoptera, Aphidiidae), Acta Entomol. Bohemoslov. 81, 81–86.
- Stephens M.J., France C.M., Wratten S.D., Frampton C. (1998) Enhancing biological control of leafrollers (Lepidoptera: Tortricidae) by sowing buckwheat (*Fagopyrum esculentum*) in an orchard, Biocontrol Sci. Techn. 8, 547–558.
- Suckling D.M., Walker J.T.S., Wearing C.H. (1999) Ecological impact of three pest management systems in New Zealand apple orchards, Agr. Ecosyst. Environ. 73, 129–140.
- Szentkirályi F., Kozár F. (1991) How many species are there in apple insect communities?: testing the resource diversity

and intermediate disturbance hypotheses, Ecol. Entomol. 16, 491–503.

- Tasin M., Demaria D., Ryne C., Cesano A., Galliano A., Anfora G., Ioriatti C., Alma A. (2008) Effect of anti-hail nets on *Cydia pomonella* behavior in apple orchards, Entomol. Exp. Appl. 129, 32–36.
- Tscharntke T., Bommarco R., Clough Y., Crist T.O., Kleijn D., Rand T.A., Tylianakis J.M., Van Nouhuys S., Vidal S. (2007) Conservation biological Control and enemy diversity on a landscape scale, Biol. Control 43, 294–309.
- Tuovinen T. (1994) Influence of surrounding trees and bushes on the phytoseiid mite fauna on apple orchard trees in Finland, Agr. Ecosyst. Environ. 50, 39–47.
- Van Emden H.F., Williams G.F. (1974) Insect stability and diversity in agro-ecosystems, Annu. Rev. Entomol. 19, 455–475.
- Van Helden M., Fargeas E., Fronzes M., Maurice O., Thibaud M., Gil F., Pain G. (2006) The influence of local and landscape characteristics on insect pest population levels in viticulture, Bull. IOBC/wprs 29, 145–148.
- Vogt H., Weigel A. (1999) Is it possible to enhance the biological control of aphids in an apple orchard with flowering strips? Bull. IOBC/wprs 22, 39–46.
- Vogt H., Weigel A., Wyss E. (1998) Aspects of indirect plant protection strategies in orchards: are flowering strips an adequate measure to control apple aphids? in: Brunnhofer V., Soldan T. (Eds.), Proc. 6th European Congress of Entomology (Book of Abstracts), Ceské Budejovice, August 23–29, 1998, Institute of Entomology, Academy of the Czech republic and University of South Bohemia, pp. 625–626.
- Westigard P.H., Flexner L.J., Vanburskirk P., Gonzalves P., Hilton R. (1990) Dispersal pattern of the twospotted spider mite from orchard groundcover into pear, Bull. IOBC/wprs 13, 53–57.
- Whalon M.E., Croft B.A. (1986) Immigration and colonization of portable apple trees by arthropod pests and their natural enemies, Crop Prot. 5, 376–384.
- Wildbolz T. (1988) Integrated pest management in Swiss apple orchards: stability and risks, Entomol. Exp. Appl. 49, 71–74.
- Witzgall P., Stelinsky L., Gut L., Thomson D. (2008) Codling moth management and chemical ecology, Annu. Rev. Entomol. 53, 503–522.
- Wyss E. (1995) The effects of weed strips on aphids and aphidophagous predators in an apple orchard, Entomol. Exp. Appl. 75, 43–49.
- Wyss E. (1996) The effects of artificial weed strips on diversity and abundance of the arthropod fauna in a Swiss experimental apple orchard, Agr. Ecosyst. Environ. 60, 47–59.
- Wyss E., Niggli U., Nentwig W. (1995) The impact of spiders on aphid populations in a strip-managed apple orchard, J. Appl. Entomol. 119, 473–478.
- Yan Y., Yu Y., Du X., Zhao B. (1997) Conservation and augmentation of natural enemies in pest management of Chinese apple orchards, Agr. Ecosyst. Environ. 62, 253–260.
- Zandstra B.H., Motooka P.S. (1978) Beneficial effects of weeds in pest management – a review, PANS 24, 333–338.
- Zehnder G., Gurr G.M., Kühne S., Wade M.R., Wratten S.D., Wyss E. (2007) Arthropod pest management in organic crops, Ann. Rev. Entomol. 52, 57–80.
- Zhang W., Ricketts T.H., Kremen C., Carney K., Swinton S.M. (2007) Ecosystem services and dis-services to agriculture, Ecol. Econ. 64, 253–260.

# **Pathogenic and Beneficial Microorganisms in Soilless Cultures**

J. Vallance, F. Déniel, G. Le Floch, L. Guérin-Dubrana, D. Blancard, and P. Rey

### Contents

1	Introduction						
2	Ecology of the Microflora in Soilless Systems						
	2.1	Influence of the Kind of Substrate					
		on Microflora	714				
	2.2	Root System and Nutrient Solution Microflora .	714				
	2.3	Influence of the Rhizosphere on the Microbial					
		Communities	714				
	2.4	Evolution of Microbial Communities	715				
3	Uniqu	e Disease Problems in Soilless Cultures	715				
	3.1	Infections by Zoosporic Oomycetes	715				
	3.2	Complex of Pathogens on Necrotic Roots	716				
	3.3	Symptomless and Minor Pathogen Infections					
		on Roots	716				
	3.4	Other Potentially Pathogenic Microorganisms					
		in Soilless Cultures	717				
4	Effect of Disinfection Techniques on the Microflora						
	of Soi	lless Systems	717				
	4.1	Active Methods	717				
	4.2	Passive Method: Slow Filtration	718				
5	Diseas	se Suppression in Soilless Systems	719				
6	Mana	gement of the Soilless Microflora for Disease					
	Suppr	ression	720				
	6.1	Increasing the Level of Suppressiveness by the					
		Addition of Antagonistic Microorganisms	720				
	6.2	Use of a Mixed Culture of Antagonistic					
		Microorganisms with Disinfection Techniques .	721				
	6.3	Nutritional Amendments	722				
7	Concl	usion	723				
References							

P. Rey (🖂)

Abstract Soilless cultures were originally developed to control soilborne diseases. Soilless cultures provide several advantages for growers such as greater production of crops, reduced energy consumption, better control of growth and independence of soil quality. However, diseases specific to hydroponics have been reported. For instance, zoospore-producing microorganisms such as *Pythium* and *Phytophthora* spp. are particularly well adapted to aquatic environments. Their growth in soilless substrates is favoured by the recirculation of the nutrient solution. These pathogenic microorganisms are usually controlled by disinfection methods but such methods are only effective as a preventive measure. Contrary to biofiltration, active treatments such as UV, heat and ozonisation have the disadvantage of eliminating not only the harmful microorganisms but also the beneficial indigenous microorganisms. Here, we review microbial populations that colonise ecological niches of hydroponic greenhouse systems. Three topics are discussed: (1) the general microflora; (2) the pathogenic microflora that are typical to hydroponic systems; and (3) the non-pathogenic and possibly beneficial microflora, and their use in the control of plant diseases in soilless greenhouse systems. Technical, economic and environmental concerns are forcing the adoption of new sustainable methods such as the use of microbial antagonists. Thus, increased attention is now focused on the role of natural microflora in suppressing certain diseases. Managing disease suppression in hydroponics represents a promising way of controlling pathogens. Three main strategies can be used: (1) increasing the level of suppressiveness by the addition of antagonistic microorganisms; (2) using a mix of microorganisms with complementary ecological traits

UMR Santé Végétale 1065, INRA, ENITA de Bordeaux, Université de Bordeaux, 33175 Gradignan, France e-mail: prey@bordeaux.inra.fr

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_31,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Vallance et al., Agron. Sustain. Dev. In press. DOI: 10.1051/agro:2010018

and antagonistic abilities, combined with disinfection techniques; and (3) amending substrates to favour the development of a suppressive microflora. Increasing our knowledge on beneficial microflora, their ecology and treatments that influence their composition will help to commercialise new, ready-to-use substrates microbiologically optimised to protect plants in sustainable management systems.

**Keywords** Antagonistic agents • Biological control • Microbial ecology • Disinfection methods • Hydroponics • Recirculating solutions • Root rots • Suppressive microflora • Wilting • Zoosporic pathogens

### 1 Introduction

Soilless cultures are used worldwide. Depending on the country, growers use a variety of complex technologies, all of which offer advantages making them appropriate alternatives to traditional soil culture (Fig. 1a). In cases where the soil is polluted by chemical residues or contaminated by pathogens which colonise and persist in the soil for years or when excessive salinity causes water problems, soilless cultures can be an alternative. The main advantage of soilless cultures is that plants grow in a controlled environment. For instance, nutrient solution supply, electrical conductivity, pH and temperature are monitored and regulated by the grower. It provides an ideal environment for growth and development of plants and a greater yield is frequently obtained than with traditional cultural methods. The majority of greenhouse crops are grown using artificial substrates (Fig. 1b), which improves control of water, aeration, nutrition and root distribution. These systems were originally developed as open systems and excess nutrient solution was disposed of outside the greenhouse. In recent years, closed hydroponic systems have been developed to minimise pollution. In a closed system, the nutrient solution is recovered, replenished with nutrients and water, depending on plant uptake, and the pH adjusted before resupplying to the plants.

Microbial contamination of the root system in these culture systems can arise from many sources: plant material, growth media, and water from lakes, rivers and wells (Stanghellini and Rasmussen, 1994). Root colonisation by fungi and bacteria is favoured by at least three factors: (i) genetically uniform host plants, (ii) environmental conditions, i.e. suitable temperature and moisture regime, and (iii) rapid dispersal of rootcolonising agents throughout the cultural system via the recycled nutrient solution.

The activity of microorganisms, however, may be pathogenic or protective, so two scenarios are possible. (1) One of the reasons for developing soilless culture was to prevent root diseases caused by soil-pathogenic microorganisms. Although a decrease in the diversity of root-infecting microorganisms has been reported, root diseases still occur frequently in hydroponics and disease outbreaks are sometimes greater than in soil (Stanghellini and Rasmussen, 1994). Some minor infections have become threats in soilless culture, indicating that unique diseases are observed with this method of plant cultivation. (2) The role of natural microflora in suppressing certain diseases was demonstrated by comparing systems with and without their original microflora (Postma et al., 2000; Minuto et al., 2007). Indeed, it has been shown that the natural microflora can suppress diseases (Berger et al., 1996; Chen et al., 1998) and that a high density of bacteria in the rhizosphere can limit pathogenic attacks on roots (Tu et al., 1999). From these observations ensued the hypothesis that indigenous bacteria were involved in disease biosuppression.

In this study we focused on the microbial communities colonising the root systems of plants growing in soilless cultures and highlighted the specificity of microbes in this type of cultivation system. Three topics were reviewed: (i) the general microflora; (ii) the pathogenic microflora of typical diseases related to hydroponics; and (iii) the non-pathogenic and possibly beneficial microflora and their use in the control of plant diseases in soilless greenhouse systems.

### 2 Ecology of the Microflora in Soilless Systems

Soon after the start of a soilless culture, a microflora rapidly colonises three ecological niches: the substrate, the nutrient solutions and the rhizosphere of the cultivated plants. The density and diversity of this microflora are affected by the type of substrate (organic or inorganic), the nutrients in the solutions and the age and cultivar of the plant species.



**Fig. 1** Tomato soilless culture and the main associated fungal pathogens. Suspended substrate in a tomato soilless culture (**a**), rockwool containing healthy and altered roots (**b**), *Phytophthora cryptogea* sporangia on the surface of a necrosed root (**c**),

*Pythium aphanidermatum* oospores (round with a thick wall) in the root cortex cells (**d**), macroconidia of *Fusarium oxysporum* f. sp. *radicis-lycopersici* with chlamydospores in formation (**e**), *Colletotrichum coccodes* acervulus with black seta (**f**)

Cultural methods have been used to characterise this microflora, but in recent years other methods based on sole-carbon-source utilisation (Khalil and Alsanius, 2001; Khalil et al., 2001b), phospholipid fatty acid profiling (Waechter-Kristensen et al., 1996; Khalil and Alsanius, 2001; Khalil et al., 2001a, b) and molecular fingerprinting (Postma et al., 2000; Calvo-Bado et al., 2003, 2006) have provided structural and functional analysis of the soilless microflora. Recent studies on microflora have provided key information on the microbial diversity and dynamics of soilless systems.

# 2.1 Influence of the Kind of Substrate on Microflora

In soilless cultures a microflora rapidly develops soon after the start of the culture via the plants and the water supply, even though inorganic substrates contain few microbes. Once plants are introduced into greenhouses, extensive colonisation of rockwool substrates by bacteria and fungi rapidly occurs (Price, 1980; Carlile and Wilson, 1991). Inorganic substrates are mainly colonised by bacteria while organic substrates are colonised by fungi (Koohakan et al., 2004). In the case of crops of tomatoes, for instance, bacteria including fluorescent pseudomonads were higher in rockwool than in peat substrates and the reverse was observed for fungi, actinomycetes and Trichoderma spp. (Khalil and Alsanius, 2001). This might be due to the presence of available organic compounds within the organic substrates which may modify the microbial equilibrium through reduced competition (Koohakan et al., 2004). The level of conduciveness to the diseases caused by a given pathogenic agent might be determined by the nature (structure, composition) of the growth substrate of the crop. For instance, rockwool is more conducive to Pythium root rot and crown rot in cucumber culture than coir dust, pumice and perlite (van der Gaag and Wever, 2005). Temperature and oxygen concentration did not explain the differences between the media but the higher incidence of disease on rockwool was associated with a much greater water content than in the three others. When the height of the rockwool slabs was increased, the percentage of diseased plants decreased. These results indicated that water content plays a major role in the development of root and stem rot and that the type and height of substrate are important tools for decreasing yield losses.

# 2.2 Root System and Nutrient Solution Microflora

Microorganisms multiply rapidly on roots and in nutrient solutions. Large populations of heterotrophic bacteria  $(10^5-10^6 \text{ cfu mL}^{-1})$  developed in the circulating nutrient solutions 20 h after planting tomatoes (Berkelmann et al., 1994). The number of bacteria on young tomato roots can be as high as  $10^{10} \text{ cfu g}^{-1}$  of fresh roots (Waechter-Kristensen et al., 1994). However, there are differences between microbial communities colonising roots and nutrient solutions; more fungi and bacteria were detected on roots than in nutrient solutions (Koohakan et al., 2004). Besides the densities, the structure and the diversity of bacterial communities, as assessed by a molecular fingerprint method (Single-Strand Conformation Polymorphism, SSCP), were also different on roots and nutrient solutions (Renault, 2007).

The cultural systems (inorganic and organic media, deep flow technique and nutrient film technique) favoured in different ways the growth of unique indigenous microorganisms (Koohakan et al., 2004). Fungi and Fusarium spp. were found to colonise preferentially roots grown in a coconut-fibre system (organic medium) compared with a rockwool system (inorganic medium). Pythium spp. were mainly detected in nutrient solutions and on roots from the nutrient film technique system. Among the non-specific bacterial genera, aerobic bacteria seemed predominant on roots and in nutrient solutions, with only slight differences between the four systems (inorganic and organic media, deep flow technique and nutrient film technique). Whatever the system, fluorescent pseudomonads were frequently detected on roots and in nutrient solutions, which was consistent with previous findings showing that 40% of the cultivable bacteria belonged to the genus Pseudomonas, known to contain potentially antagonistic agents toward pathogens (Berkelmann et al., 1994). Similar results were obtained in the recycled nutrient solution during a six-month experiment in a soilless tomato greenhouse (Déniel et al., 2004). These findings might be explained by the fact that the temperature, high nitrogen content and oxygen concentration of the nutrient solutions offer an optimal growth environment for this genus.

# 2.3 Influence of the Rhizosphere on the Microbial Communities

There is a clear relationship between cultivated plants and the establishment of the rhizosphere microflora. In closed hydroponic systems, it results from the release of organic compounds by the roots (Waechter-Kristensen et al., 1997). Passive or active leakage of carbon sources from plant roots differs in quantity and quality depending on plant species, plant cultivar and environmental factors such as light, climate, nutrient source, pH, humidity, etc. Whatever the hydroponic habitat, the diversity of microorganisms depends on their ability to metabolise the available carbon sources. Although a nutrient film technique system is much simpler than a soil-based culture system, SSCP analyses showed the bacterial diversity of the rhizoplane to be as high as that of the rhizosphere in soil (Chave et al., 2008). However, further studies comparing the microorganisms colonising soil and soilless cultures are needed to draw any conclusion.

### 2.4 Evolution of Microbial Communities

As mentioned above, biological processes in the rhizosphere are strongly affected by plant root exudates that attract specific microbial populations and stimulate their growth and evolution. Based on viable counts, aerobic bacteria colonising the rhizosphere of four types of soilless tomato production systems (inorganic substrate: rockwool; organic substrate: coconutfibre; deep flow technique, nutrient film technique) were found to become stable at  $10^{10}$  cfug<sup>-1</sup> (of fresh roots) in all systems investigated, contrary to fungi, that tended to increase throughout the experiment (Koohakan et al., 2004). However, changes in the composition of the microflora have been demonstrated by molecular and biochemical analyses. For instance, Khalil et al. (2001b) highlighted the differences between the microflora of two supposedly identical hydroponic cultivations by comparing solecarbon-source utilisation (SCSU) patterns and phospholipid fatty acid profiles (PLFA). In tomato soilless cultures, Renault et al. (2008) also observed a temporal shift over a cropping season in the bacterial composition both in the nutrient solution and on the roots. Indeed, community-level physiological profiles (CLPPs) indicated that bacterial metabolism in nutrient solutions progressively shifted from carbohydrates towards the degradation of specific amino acids and carboxylic acids.

There is no consensus about whether shifts in the rhizosphere microflora can result from pathogenic attacks. Indeed, changes in the microbial communities of the rhizosphere could be a consequence of both root damage caused by pathogens such as P. ultimum and secondary colonisation due to the resulting nutrient leakage (Naseby et al., 2000; Hagn et al., 2008). On the other hand, it has been reported that the microbial communities established early on the roots of tomatoes grown in soilless systems were robust and resistant to the effect(s) of the introduction of oomycete pathogens or of switching from a recirculating to a run-to-waste nutrient supply (Calvo-Bado et al., 2006). However, this assumption, arising from experiments conducted over only 1.5 months, is contradicted by the observation of changes in the microbial communities of tomato plants grown hydroponically over the 6-month experiments of Vallance et al. (2009). SSCP analyses of three different DNA regions indicated increases in the complexity and size of the fungal microflora as the cropping season progressed. Nevertheless, both studies suggest that there are no substantial changes in the genetic structure of the indigenous rhizospheric fungal community after root inoculation with the non-pathogenic oomycete P. oligandrum or the pathogenic oomycetes Pythium group F, P. aphanidermatum and P. cryptogea.

# 3 Unique Disease Problems in Soilless Cultures

#### 3.1 Infections by Zoosporic Oomycetes

Among the pathogenic microorganisms frequently detected in hydroponic cultures, those producing zoospores, i.e. Pythium spp. and Phytophthora spp., are particularly well adapted to these cultivation systems (Favrin et al., 1988; Rafin and Tirilly, 1995) (Figs. 1c, 1d). As zoospores can swim, recycling facilitates rapid dissemination and subsequent root infection of the whole culture (McPherson et al., 1995). Disease epidemics can occur, particularly in periods of stress, because of high temperatures and the low concentrations of dissolved oxygen in the nutrient solution (Gold and Stanghellini 1985; Stanghellini and Rasmussen 1994; Chérif et al., 1997). Highly pathogenic Pythium species, i.e. Pythium ultimum, P. irregulare and P. aphanidermatum (Blancard et al., 1992; Jenkins and Averre, 1983; Linde et al., 1994), caused root rot and wilting.

In Brittany (France), two stages in root infection by Pythium spp. in commercial tomato greenhouses were observed by Rey et al. (2001). The first is generally from the start of the winter crop (February) to June. A small population of Pythium spp. is frequently detected. The population then dramatically increases between July-August and the end of the cropping season (October-November); this increase is sometimes associated with root necrosis and root rot, but generally infections are limited to root necroses and are even symptomless. This pattern was particularly observed in greenhouses with organic (peat) and, to a lesser extent, inorganic substrates (rockwool). With a nutrient film technique system, Pythium spp. invasion was earlier and more severe than in other cultures, but with no amplification of symptoms.

A DNA macroarray for the detection and identification of more than 100 Pythium species was developed to assess the number and diversity of Pythium species on a single root sample (Tambong et al., 2006). This technology has the advantage of combining DNA amplification with the screening capability of DNA arrays, resulting in a high degree of sensitivity and multiple species specificity. The results of the DNA array tests confirmed that the substrate was almost free of oomycetes at the start of the plant culture. P. dissotocum (or Pythium group F) was spontaneously detected on roots throughout the growing period but other Pythium species (P. intermedium, P. ultimum and P. sylvaticum) were sporadically detected (Le Floch et al., 2007). The relative predominance of P. dissotocum (or Pythium group F) and the low diversity of Pythium species confirm the results of previous studies conducted in soilless cultures (Herrero et al., 2003; Moorman et al., 2002; Moulin et al., 1994; Rafin and Tirilly, 1995; Rey et al., 1997).

# 3.2 Complex of Pathogens on Necrotic Roots

A variety of fungal complexes and oomycetes are responsible for root necroses. A three-year experiment in tomato soilless cultures in France revealed that the distribution of the fungi and of the oomycetes was regiondependent (Blancard, unpubl. data). In the South-West, between two and five fungi and oomycetes were frequently found on roots, whereas in the five other regions (Brittany, the Eastern Pyrenees, Nantes region, Orleans region, the South-East), up to three different microorganisms were isolated from the samples. Some fungi, including *Fusarium oxysporum* f. sp. *radicis lycopersici*, and oomycetes, such as *Pythium* species, were found in all the greenhouses investigated in the six French regions (Fig. 1e). Other fungi, i.e. *Colletotrichum coccodes, Rhizoctonia solani* and *Thielaviopsis basicola*, or oomycetes such as *Phythophthora* spp. were only found on roots in some of the greenhouses (Fig. 1f).

# 3.3 Symptomless and Minor Pathogen Infections on Roots

Asymptomatic root colonisation in hydroponic cultures can be correlated with yield loss (Rey et al., 1997; Stanghellini and Kronland, 1986). Pythium dissotocum caused yield reductions of up to 54% in hydroponically grown lettuce although there was no visible damage (Stanghellini and Kronland, 1986). Such infections might be more common in soilless greenhouse systems than originally thought, because of the lack of root symptoms (Favrin et al., 1988). Immunoenzymatic staining procedures showed that Pythium spp. were the most frequent fungal invaders in asymptomatic roots of hydroponically grown tomato plants. Pythium spp. represented around 40% of the colonised segments as opposed to 12% for the other fungi. Pythium group F accounted for 75 to 90% of all the Pythium isolates from the loose or dense mycelia of Pythium spp. on the root epidermis (Rafin and Tirilly, 1995; Rey et al., 1997). Certain strains produce large numbers of zoospores (Rafin, 1993), possibly facilitating the spread and the development of Pythium group F in soilless cultures. When plants were grown under optimal conditions Pythium group F-infected roots were symptomless. However, roots looked generally macroscopically healthy but the oomycete caused limited changes in the root cortex (Rey et al., 1998) and produced metabolites that may facilitate Pythium group F infections (Rey et al., 2001). Moreover, due to high Pythium group F populations over the cropping season, limited but repeated damage to root cortexes could lead to slight yield reductions (Rey et al., 1997). Severe

damage, such as root rot, only occurs when plants are placed under physiological stress conditions, i.e. lack of oxygen in nutrient solutions (Chérif et al., 1997). The nature of Pythium group F is still unclear. The taxonomic position of this oomycete has only become clearer in recent years with the increased interest in Pythium group F. Van der Plaats-Niterink (1981) used the term group F because oomycetes of this group only produce non-inflated filamentous sporangia on traditional culture media and sexual structures are not observed. However, after molecular characterisation of Pythium group F isolates by ribosomal and intermicrosatellite-DNA regions analysis, Vasseur et al. (2005) suggested that Pythium group F isolates could be P. dissotocum-like isolates unable to form sexual structures on traditional media. Moreover, Lévesque and De Cock (2004) suggested that Pythium group F could be related to P. dissotocum because of the complete homology of ITS sequences.

### 3.4 Other Potentially Pathogenic Microorganisms in Soilless Cultures

The pathogenicity of a few microorganisms (Humicola sp., Olpidium brassicae and Plectosporium tabacinum) (Figs. 2a-2e) needs to be determined because some root microorganisms of minor importance in soils have become of major economic importance in hydroponic cultures (Stanghellini and Rasmussen, 1994). Hydroponics, for instance, have favoured the development of Phytophthora cryptogea on lettuce, whereas, in the field, no attacks by this fungus have been reported. Plectosporium tabacinum (formerly Fusarium tabacinum), frequently isolated from soilless tomato cultures in France (Blancard, unpublished data) is a possible pathogen. It is detected on a variety of soil-grown plants, i.e. melon (Soran and Ozel, 1985), sunflower (Mirza et al., 1995) and basil (Minuto et al., 1997). Matta (1978) and Pascoe et al. (1984) reported that it caused necrotic lesions on young leaves in tomato plants and El-Gindy (1991) noticed necrosis and root rot in plantlets. Such symptoms have never been observed on tomato plants grown hydroponically. However, considering the pathogenic potential of P. tabacinum and its frequency in greenhouses, its pathogenicity in hydroponics needs to be assessed. Another example is *Humicola fuscoatra*. Gruyter et al. (1992) reported the association of *H. fuscoatra* with corky root symptoms in wilted glasshouse tomatoes. However, Menzies et al. (1998) pointed out that *Humicola fuscoatra* colonised roots, but did not cause necrosis and was, therefore, not pathogenic in tomato plants. These findings highlight the difficulties in distinguishing minor pathogens from other fungi, as both frequently colonise roots in soilless cultures.

# 4 Effect of Disinfection Techniques on the Microflora of Soilless Systems

Closed hydroponic systems increase the risk of pathogen attack by using water contaminated with pathogenic microorganisms (McPherson et al., 1995; van Os et al., 1999). Therefore, prevention of these infections has become a major challenge in the last decade (Runia, 1995; Ehret et al., 2001).

### 4.1 Active Methods

The so-called "active" methods disinfect the nutrient solutions and are very effective (Ehret et al., 2001; Goldberg et al., 1992; Rey et al., 2001; Runia, 1995; Steinberg et al., 1994); for example, UV radiation and heat treatment can eliminate up to 99% of the microflora colonising the flowing solutions. UV irradiation of recirculating solution was effective in controlling Pythium spp.-induced root rot in tomato and cucumber plants (Postma et al., 2001; Zhang and Tu, 2000). Tirilly and Letard (1997) reported a delay in Pythium root infection in soilless culture with this method; however, in several cases there was no difference in root colonisation from non-disinfected greenhouses. Moreover, re-contamination of the disinfected nutrient solution nullified the effect of disinfection (Déniel et al., 1999). Such drastic treatments create a microbiological vacuum in which microbial pioneers spread more easily because of the lack of competition (Paulitz and Bélanger, 2001; Postma, 2004). The microbial differences in solutions treated with UV and slow filtration often disappeared once they flowed through the rockwool slabs containing plant roots (van Os et al., 2004). Chlorination is effective in disinfecting water in storage tanks and reduces and



**Fig. 2** In situ and in vitro appearance of three fungi sometimes associated with root rot in tomato soilless culture but whose aggressiveness has never been proven on this Solanaceae. *Olpidium brassicae* resting spores aligned in several root cells (**a**), *Plec*-

delays root colonisation by *Pythium* spp. (Déniel et al., in press). However, this treatment has the disadvantage of eliminating not only harmful but also beneficial indigenous microorganisms; a weakness of "active" methods of disinfection. Zhang and Tu (2000) imputed the lack of control of *P. aphanidermatum* on tomato roots to the reduction of bacterial communities caused by UV radiation.

*tosporium tabacinum* bicellular conidia within root cortex cells (b), phialides of *Plectosporium tabacinum* (c), aleuriospores (dark brown) of *Humicola* sp. (d), phialides of *Humicola* sp. perpendicular to the mycelium; they form chains of conidia (e).

### 4.2 Passive Method: Slow Filtration

The traditional technique of slow filtration, used for more than 100 years for water disinfection (Graham and Collins, 1996; Ellis, 1985), has been adapted for horticulture over the last decade (Ehret et al., 2001). Water flows slowly through a bed of substrate, i.e. sand, rockwool or pozzolana; mechanical and biological factors are thought to be responsible for the efficacy of the system (Ellis, 1985; Weber-Shirk and Dick, 1997). Experiments to improve slow filtration efficacy have focused on the determination of flow rates through the filter unit as well as on the nature and the optimal depth of substrates in filter tubes (Wohanka et al., 1999). Further investigations showed that the formation of bacterial microcolonies or biofilms on substrates enhanced efficiency. Indeed, after sterilising a filtering column, a dramatic loss in Xanthomonas campestris pv. pelargonii elimination has been reported (Brand and Wohanka, 2001). Pseudomonas was the predominant genus (50%) from the cultivable bacteria colonising the filtering media, especially the top layers of sand filters, and 10% of isolates were identified as Bacillus (Brand, 2000; Calvo-Bado et al., 2003). The Bacillus and Pseudomonas genera were recently reported to account for 42 to 86% of the total cultivable bacterial flora in a biocenosis film of pozzolana grains used as filtering medium (Déniel et al., 2004).

Pathogens eliminated efficiently by this technique include zoosporic fungi, i.e. Phytophthora spp., bacteria, i.e. Xanthomonas campestris, nematodes and even viruses (Ehret et al., 2001; van Os et al., 1999). During a 3-year experiment in a commercial greenhouse, Déniel et al. (2006) reported that a biofilter eliminated more fungi than bacteria under tomato production conditions. The efficiency of elimination of pathogenic fungi was genus-dependent. Pythium spp. were more effectively eliminated (99%) than Fusarium oxysporum (92.7 to 99.3%). The high percentage of Pythium spp. elimination was correlated with low root colonisation by these pathogens. Effluents of filtering columns have been shown to be colonised by a considerable natural bacterial microflora  $(10^2 - 10^4 \text{ cfu mL}^{-1})$ (Déniel et al., 2004, 2006; Renault, 2007). Moreover, molecular fingerprinting analyses of the total microflora (denaturing gradient gel electrophoresis, DGGE, and SSCP) pointed out clear changes in bacterial communities after the passage of the nutrient solution through the filter unit (Postma et al., 1999; Renault, 2007). Thus, slow filtration preserved part of the natural microflora, because it is harmless to specific groups of bacteria which are assumed to preserve microbial ecosystems in the plant rhizosphere. Furthermore, resident bacteria of nutrient solutions were shown to reduce Pythium root rot in closed

soilless systems (Tu et al., 1999). The potential benefit of microflora in soilless cultures thus has to be taken into account.

# 5 Disease Suppression in Soilless Systems

Pathogen-suppressive soils have been defined as "soils in which (i) the pathogen does not establish or persist; (ii) establishes but causes little or no damage; or (iii) establishes and causes disease for a while but thereafter the disease is less important, although the pathogen may persist in the soil" (Borneman and Becker, 2007). Soils suppressive to several pathogens have been widely described and investigated (Alabouvette et al., 1979; Jager et al., 1979; Lifshitz et al., 1984; Garibaldi et al., 1989; Whipps and Lumsden, 1991), while the first studies of suppressiveness in soilless systems were by McPherson et al. (1995) and Tu et al. (1999). Both studies demonstrated the potential of the indigenous microflora to inhibit root diseases in hydroponic cultures. In soilless cultures, the term "suppressiveness" referred to the cases where (i) the pathogen does not establish or persist; or (ii) establishes but causes little or no damage. McPherson et al. (1995) described the spread of Phytophthora cryptogea in tomato nutrient film technique systems. In closed systems, the pathogen caused less damage than in the parallel run-to-waste ones; they therefore suggested that the potentially beneficial microflora colonising the recycled nutrient solution were responsible for disease suppression. They also suggested that the method of disinfection, i.e. "active" or "passive" (by total or partial elimination of the microflora) could be important in the maintenance of the disease suppression. Tu et al. (1999) also showed that Pythium root rot disease was less severe in closed rockwool systems than in open culture due to the greater numbers of bacteria in closed systems. They found a strong correlation between the resident bacteria and the biosuppression of Pythium.

The presence of microflora suppressing *Pythium* aphanidermatum in cucumber rockwool substrate has been reported and some of the microorganisms involved in the suppressiveness identified (Postma et al., 2000, 2004, 2005). *Pythium* damage was lower in
non-autoclaved than in autoclaved rockwool; the disease incidence was reduced by 50 to 100%. Suppressiveness could be restored in sterilised rockwool substrates by re-introducing the original microflora through contact with untreated rockwool or through the nutrient solution taken from untreated slabs. These results indicate that disease suppression is of biological origin and is transferable. Experiments on the microbial communities of rockwool showed a positive association between disease suppressiveness and the composition and diversity of bacteria and culturable filamentous actinomycetes. Actinomycetes may prevent the colonisation of dead root fragments by Pythium zoospores, whilst bacteria may secrete antibiotics, surfactants, etc. preventing colonisation of fresh root fragments.

Suppression of *Fusarium oxysporum* f. sp. *radicis lycopersici* has also been demonstrated. The incidence of *Fusarium oxysporum* f. sp. *radicis lycopersici* on tomato seedlings was significantly reduced with recycled, non-disinfected rockwool compared with new rockwool (Minuto et al., 2007); and in tomato soilless culture, by the re-use of perlite and perlite-peat substrates (Clematis et al., 2008). The indigenous microorganisms colonising these recycled substrates were considered responsible for the suppressive effects.

the suppressive microflora How becomes established is relatively unknown, but it has been suggested that pathogens themselves might influence suppressiveness. For instance, a study showed that P. ultimum induced shifts in cucumber indigenous microflora, favouring groups known to include potential biocontrol agents (Hagn et al., 2008). However, knowledge of structural and functional interactions and synergisms between the microorganisms of the suppressive microflora is limited and the influence of the plant and the pathogens on the whole system needs further investigation (Weller et al., 2002; Burdon et al., 2006).

## 6 Management of the Soilless Microflora for Disease Suppression

Factors influencing disease suppression such as the activity of the total microflora, the diversity of the microbial communities and the presence of specific antagonists are not fully understood (Postma, 2004). Nevertheless, managing disease suppression in hy-

droponics represents a promising way of controlling pathogens. Three main strategies can be used: (i) increasing the level of suppressiveness by the addition of antagonistic microorganisms; (ii) using a mixed culture of microorganisms with complementary ecological traits and antagonistic abilities combined with disinfection techniques; and (iii) amending substrates to favour the development of the suppressive microflora.

## 6.1 Increasing the Level of Suppressiveness by the Addition of Antagonistic Microorganisms

Environmental conditions in greenhouses are controlled and can be optimised to suit antagonistic agents. The biological vacuum and the limited volume of the matrix of the soilless substrates are thought to facilitate the introduction, establishment and interaction of the biocontrol agent with the root environment (Paulitz and Bélanger, 2001; Postma, 2004). Thus, representatives of a range of bacterial (Pseudomonas, Burkholderia, Bacillus, Serratia, Actinomycetes), fungal (Trichoderma, Penicillium, Gliocladium, non-pathogenic Fusarium) and oomycete (Pythium) groups have been tested as biocontrol agents in soilless cropping systems. The antagonistic activities of these microorganisms can be divided into several categories: competition for nutrients and space, parasitism, antibiosis and systemic induced resistance (Garbeva et al., 2004; Alabouvette et al., 2006; Lemanceau et al., 2006). Nevertheless, biocontrol of root diseases is often inefficient and only a few antagonists are available commercially.

The lack of efficiency is due to unsuitable methods of selection of antagonistic microorganisms. Results from in vitro studies did not always correlate with the antagonistic activity of the biocontrol agent once they were introduced into greenhouses (Fravel, 2005; Alabouvette et al., 2006; Georgakopoulos et al., 2002). These results also demonstrated the importance of the medium used for doing the in vitro tests; it has to be as close as possible to the environment into which the antagonists will be introduced. Even then, the colonisation, survival and antagonistic activity of the biocontrol agent may be insufficient and/or inconsistent at the infection site because the antagonist is not adapted to the soilless environment. The use of microorganisms selected from the indigenous suppressive microflora and not from a suppressive soil or a different crop might solve this problem: the microorganisms would be better adapted to the soilless crop environment and the ecological niche where their interaction with the pathogens will take place.

For example, the pathogenic fungi or oomycetes most frequently involved in root diseases in soilless cultures are those producing zoospores, such as Pythium spp. and Phytophthora spp., making them particularly well adapted to the aquatic environment of hydroponics. The use of an antagonist belonging to the same taxonomic group (i.e. oomycetes), with the same life cycle and similar properties, is of particular interest. An example of such an antagonist is the oomycete P. oligandrum (Rey et al., 2008; Vallance et al., 2009); it has been widely reported as an effective biocontrol agent (Foley et al. 1986; Jones and Deacon, 1995; Benhamou et al., 1997; Rey et al., 1998, 2005; Wulff et al., 1998). The beneficial effects of P. oligandrum are due to its potential to colonise roots without damaging the host plant cells and to survive in the rhizosphere. P. oligandrum biocontrol in the rhizosphere is a complex process including direct control of pathogens by mycoparasitism, antibiosis or competition for nutrients and space; and/or indirect control via the plant, i.e. induction of resistance and growth promotion (Le Floch et al., 2005; Rey et al., 2008). Persistent root colonisation by P. oligandrum strains may be associated with an increase in tomato yield in soilless cultures (Le Floch et al., 2003), a transient increase (Le Floch et al., 2007) or not (Vallance et al., 2009).

When root colonisation by P. oligandrum is assessed, results from molecular (DNA macroarray and real-time PCR) and culture-dependent methods may be contradictory. Indeed, in the experiment of Le Floch et al. (2007), P. oligandrum was detected throughout the growing season (6 months) with molecular methods, but only for three months with plate counting on semi-selective media. These findings have important implications for biocontrol strategies aimed at protecting plants. Indeed, two different strategies could be envisaged: (i) based on cultural data, P. oligandrum inoculation on roots should be repeated three months after the first application; or (ii) conversely, based on molecular results, reinoculation is unnecessary because P. oligandrum is still present. In conclusion, the second strategy probably represents the true pattern of root colonisation by the antagonist, because detection by DNA array and real-time PCR is more accurate. Appropriate methods should therefore be used to detect the antagonistic agent(s) in assessment of biocontrol.

A strategy for increasing suppressiveness and therefore making biocontrol more successful might be to associate several antagonistic agents with complementary and/or synergistic modes of action against one or several pathogens (Spadaro et al. 2005). This is the case in naturally suppressive soils, where suppression is the result of complex interactions between several microorganisms acting together. Known examples are soils suppressive to Fusarium wilts where non-pathogenic Fusarium and fluorescent Pseudomonas were identified as the main antagonists (Alabouvette and Lemanceau, 1999). The nonpathogenic Fusarium competes for carbon sources while bacterial antagonists produce siderophores competing for iron. In soilless cropping systems, the association of the non-pathogenic Fusarium strain Fo47 and fluorescent Pseudomonas strain C7R12 controlled fusarium diseases better than single inoculations of each antagonistic microorganism (Eparvier et al., 1991). Another strategy was to combine inoculation of Lysobacter enzymogenes with chitosan. Chitosan enhanced the biocontrol efficacy of L. enzymogenes in the control of P. apahidermatum in cucumber soilless greenhouse systems. Chitosan either served as a C and N source for the antagonist, induced antagonistic gene expression, or both (Postma et al., 2009).

## 6.2 Use of a Mixed Culture of Antagonistic Microorganisms with Disinfection Techniques

A more complex strategy consists of combining nutrient solution disinfection methods with biocontrol agents to colonise and protect the roots from pathogenic attack. One of the first experiments of this type combined slow filtration and *P. oligandrum* inoculation on roots in a tomato soilless greenhouse system (Rey et al., 1999). Then, the association of slow sand filtration and antagonistic strains of *Fusarium* spp. and *Trichoderma* spp. isolated from a gerbera rhizosphere was successfully tested (Garibaldi et al., 2003). A similar experiment also reported that slow filtration and antagonistic fungi (*Fusarium* spp. and *Trichoderma* spp.) operated synergistically to significantly reduce the incidence of *P. cryptogea* root

rot in gerbera crops (Grasso et al., 2003). Another strategy with slow filtration is to enhance efficiency by biological activation of the filtering columns with bacteria with suppressive traits, i.e. antagonistic activities, or siderophore and auxin production (Déniel et al., 2004). These bacteria, i.e. Bacillus and Pseudomonas strains, were isolated from a mature tomato hydroponic slow filtration unit and then inoculated into a new filter (Renault et al. 2007). Further investigations showed that the six-month period for the control filter to reach maximum efficiency against F. oxysporum was shortened in the bacteria-amended filter; in addition, filtration was highly efficient from the first month. Fast colonisation of pozzolana grains by selected bacteria and their subsequent interaction with F. oxysporum is probably responsible for filter efficiency. Pseudomonas spp. are supposed to act by competing for nutrients and *Bacillus* spp. by antibiosis and/or direct parasitism (Déniel et al., 2004). However, after nine months of operation, bacteria from the genera Pseudomonas and Bacillus, used to inoculate the filters, were not recovered in significant numbers from substrates in these filtering columns (Renault, 2007). Therefore, although early bacterial inoculations promote filter efficacy and induce a significant shift in microbial communities, the inoculated bacteria do not colonise the filtering substrates for long periods.

#### 6.3 Nutritional Amendments

Although physico-chemical factors influence the prevalence of Pythium diseases in certain substrates (van der Gaag and Wever, 2005), the main factor regulating disease suppression in hydroponic cultures is the microflora. The rhizosphere competency of potential biocontrol agents is often limited due to a lack of available organic nutrients in soilless growth media. Indeed, the main source of nutrients for the microflora on inorganic substrates is the plant roots, i.e. exudates, mucigel, sloughed root cells, etc. In conventional agriculture many other sources are available: organic amendments such as compost can be used as fertilisers or to improve the physical structure of the soil. Composted organic amendments are also substrates capable of suppressing plant diseases caused by a wide range of pathogens and pests, including bacteria, fungi and nematode species (Hoitink and Boehm,

1999; Alabouvette et al., 2006; Termorshuizen et al., 2006). Therefore, to maintain a critical threshold population of antagonistic microorganisms in soilless substrates, two approaches (similar to those in conventional cultures) could be considered: (a) the use of a different organic material, i.e. compost, as an alternative substrate for greenhouse production, and (b) the introduction of a food base for the biocontrol agent to sustain its antagonistic activity without stimulating that of the pathogen.

(a) Composted organic amendments have been tested as alternative substrates to peat in soilless systems to preserve peat bogs. Two different types of citrus compost and their water extracts were investigated as partial peat substitutes for melon seedlings in greenhouse nurseries. Compared with peat, both composts (containing plant nutrients and auxin- and cytokinin-like compounds) enhanced the plant growth; biocontrol of Fusarium oxysporum was also achieved due to the biotic component. Water extracts had no effect on plant yield but their biocontrol ability was similar to that of their solid matrices (Bernal-Vicente et al., 2008). Another study showed that the suppressiveness of compost is related to the ability of its microflora to degrade organic compounds. The microbial communities associated with three substrates with varying capacities of Fusarium wilt suppression were characterised: peat (conducive to wilt), cork (moderately suppressive) and grape marc (very suppressive). The nature and composition of the plant growth medium determined the microbial communities: in suppressive media, the microflora preferentially metabolised less easily biodegradable compounds such as carboxylic acids, amino acids, amines, phenolic compounds and polymers; while the microflora of peat used mostly sugars (Borrero et al., 2006).

(b) As the availability of nutrients is a limiting factor for the growth of the microbial communities in various plant habitats, the use of nutritional amendments has been studied to selectively increase the communities' size and the biocontrol efficacy of a target biocontrol agent. The feasibility of selective enhancement and maintenance of desired populations of naturallyoccurring biocontrol agents such as *Pseudomonas putida* by amending the nutrient solution with a nitrogen stabiliser, N-Serve<sup>®</sup>, has been demonstrated. Both active and inert ingredients in N-Serve<sup>®</sup> were involved in the suppression of root disease of pepper and cucumber caused by *Phytophthora capsici* and *P. aphanidermatum.* Xylene and 1,2,4-trimethylbenzene, the constituents of the inert fraction of N-Serve<sup>®</sup>, served as carbon sources for the selective enhancement of the pseudomonad populations, and nitrapyin, the active ingredient, reduced the vegetative growth of both pathogens (Pagliaccia et al., 2007, 2008).

#### 7 Conclusion

The last three decades have convincingly shown that, in soilless culture, the initial goal of growing plants free of soilborne microorganism attacks was not realistic. Diseases specific to this type of cultivation have been frequently reported; indeed, the elimination of the soil did not remove the pathogenic issue but has simply moved it. For instance, in comparison with soil, some diseases are only observed or have taken on a greater importance in soilless cultures. In that context, control methods have to be adapted to soilless greenhouses. One of the main options that has gradually emerged in recent years has been the use of nonpathogenic microflora. This assumption was based on the finding that if hydroponics is a "solution" for the development and spread of pathogenic zoosporic fungi and oomycetes, much evidence indicates that it can also be one for the management of the plant protective microflora. The development of sustainable control methods such as classical biological control but also new kinds of experiments, i.e. the re-use of substrates (with their suppressive microflora) or the use of suppressive ready-to-use substrates, is a must for soilless cultures. As numerous environmental parameters are controlled, managing the microflora is much easier in soilless culture than on soil. It will be a testing ground on which the results could be used for transfer to more complex systems such as soil.

#### References

- Alabouvette C., Lemanceau P. (1999) Joint action of microbials for disease control, in: Hall F.R., Menn J.J. (Eds.), Methods in biotechnology 5, Biopesticides: use and delivery, Humana Press Inc, pp. 117–135.
- Alabouvette C., Olivain C., Steinberg C. (2006) Biological control of plant diseases: the European situation, Eur. J. Plant Pathol. 114, 329–341.

- Alabouvette C., Rouxel F., Louvet J. (1979) Characteristics of *Fusarium* wilt-suppressive soils and prospects for their utilization in biological control, in: Schippers B., Gams W. (Eds.), Soil-borne Plant Pathogens. Academic Press, New-York, pp. 165–182.
- Benhamou N., Rey P., Chérif M., Hockenhull J., Tirilly Y. (1997) Treatment with the mycoparasite *Pythium oligandrum* triggers induction of defense-related reactions in tomato roots when challenged with *Fusarium oxysporum* f. sp. radicislycopersici, Phytopathology 87, 108–122.
- Berger F., Li H., White D., Frazer R., Leifert C. (1996) Effect of pathogen inoculum, antagonist density, and plant species on biological control of *Phytophthora* and *Pythium* damping-off by *Bacillus subtilis* Cot 1 in high-humidity fogging glasshouses, Phytopathology 86, 428–433.
- Berkelmann B., Wohanka W., Wolf G.A. (1994) Characterization of the bacterial flora in circulating nutrient solutions of a hydroponic system with rockwool, Acta Hort. 361, 372–381.
- Bernal-Vicente A., Ros M., Tittarelli F., Intrigliolo F., Pascual J.A. (2008) Citrus compost and its water extract for cultivation of melon plants in greenhouse nurseries. Evaluation of nutriactive and biocontrol effects, Bioresource Technol. 99, 8722–8728.
- Blancard D., Rafin C., Chamont S., Tirilly Y., Jailloux F. (1992) Phénomène de perte de racines en culture hors-sol. Rôle des *Pythiums* spp., P. H. M. Rev. Hort. 329, 35–45.
- Borneman J., Becker J.O. (2007) Identifying microorganisms involved in specific pathogen suppression in soil, Annu. Rev. Phytopathol. 45, 153–172.
- Borrero C., Ordovás J., Trillas M.I., Avilés M. (2006) Tomato *Fusarium* wilt suppressiveness. The relationship between the organic plant growth media and their microbial communities as characterised by Biolog<sup>®</sup>, Soil Biol. Biochem. 38, 1631–1637.
- Brand T. (2000) Beurteilung der bakterienflora von langsamfiltern als biologishe filtrationskomponente in geschlossenen bewässerungssystemen des gartenbaus, Geisenheimer berichte 41, Ph.D. thesis, Technischen Universität München, Geisenheim, 112 p.
- Brand T., Wohanka W. (2001) Importance and characterization of the biological component in slow filters, Acta Hort. 554, 313–321.
- Burdon J.J., Thrall P.H., Ericson L. (2006) The current and future dynamics of disease in plant communities, Annu. Rev. Phytopathol. 44, 19–39.
- Calvo-Bado L.A., Petch G., Parsons N.R., Morgan J.A.W., Pettitt T.R., Whipps J.M. (2006) Microbial community responses associated with the development of oomycete plant pathogens on tomato roots in soilless growing systems, J. Appl. Microbiol. 100, 1194–1207.
- Calvo-Bado L.A., Pettit T.R., Parsons N., Petch G.M., Morgan J.A.W., Whipps J.M. (2003) Spatial and temporal analysis of the microbial community in slow sand filters used for treating horticultural irrigation water, Appl. Env. Microbiol. 69, 2116–2125.
- Carlile W.R., Wilson, D.P. (1991) Microbial activity in growing media – a brief review, Acta Hort. 294, 196–206.
- Chave M., Dabert P., Brun R., Godon J.J., Poncet C. (2008) Dynamics of rhizoplane bacterial communities subjected to physicochemical treatments in hydroponic crops, Crop Prot. 27, 418–426.

- Chen C., Bélanger R.R., Benhamou N., Paulitz T.C. (1998) Induced systemic resistance (ISR) by *Pseudomonas* spp. impairs pre- and post-infection development of *Pythium aphanidermatum* on cucumber roots, Eur. J. Plant Pathol. 104, 877–886.
- Chérif M., Tirilly Y., Bélanger R.R. (1997) Effect of oxygen concentration on plant growth, lipidperoxidation, and receptivity of tomato roots to *Pythium* F under hydroponic conditions, Eur. J. Plant Pathol. 103, 255–264.
- Clematis F., Minuto A., Gullino M.L., Garibaldi A. (2008) Suppressiveness to Fusarium oxysporum f. sp. radicis lycopersici in re-used perlite and perlite-peat substrates in soilless tomatoes, Biological Control. 48, 108–114.
- Déniel F., Rey P., Chérif M., Guillou A., Tirilly Y. (2004) Indigenous bacteria with antagonistic and plant-growth-promoting activities improve slow-filtration efficiency in soilless cultivation, Can. J. Microbiol. 50, 499–508.
- Déniel F., Renault D., Tirilly Y., Barbier G., Rey P. (2006) Dynamic biofiltration in tomato soilless greenhouse: evolution of microbial communities on filtering media and control of potentially suppressive and pathogenic microorganisms, Agron. Sustain. Dev. 26, 185–193.
- Déniel F., Rey P., Tirilly Y. (1999) Cultures hors-sol: désinfection des solutions recyclées, Fruits Leg. 172, 73–75.
- Déniel F., Vallance J., Barbier G., Le Quillec S., Benhamou N., Rey P. (2010) Control of *Pythium* spp. root colonization in tomato soilless culture through chlorination of water storage tank, Acta Hort, in press.
- Ehret D.L., Alsanius B., Wohanka W., Menzies J.G., Utkhede R. (2001) Disinfestation of recirculating nutrient solutions in greenhouse horticulture, Agronomy 21, 323–339.
- El-Gindy A. (1991) A new root disease of tomato in Egypt caused by *Fusarium tabacinum*, Zent. Bl. Mikrobiol. 146, 77–79.
- Ellis K.V. (1985) Slow sand filtration, Crit. Rev. Environ. Control. 15, 315–354.
- Eparvier A., Lemanceau P., Alabouvette C. (1991) Population dynamics of non-pathogenic *Fusarium* and fluorescent *Pseudomonas* strains in rockwool, a substratum for soilless culture, FEMS Microbiol. Ecol. 86, 177–184.
- Favrin R.J., Rahe J.E., Mauza B. (1988) Pythium spp. associated with crown rot of cucumbers in British Columbia greenhouses, Plant Dis. 72, 683–687.
- Foley M.F., Deacon J.W. (1986) Susceptibility of *Pythium* spp. and other fungi to antagonism by the mycoparasite *Pythium oligandrum*, Soil Biol. Biochem. 18, 91–95.
- Fravel D.R. (2005) Comercialization and implementation of biocontrol, Annu. Rev. Phytopathol. 43, 337–359.
- Garbeva P., van Veen J.A., van Elsas J.D. (2004) Microbial diversity in soil: selection of microbial populations by plant and soil type and implications for disease suppression, Annu. Rev. Pythopathol. 42, 243–270.
- Garibaldi A., Guglielmone L., Gullino M.L. (1989) Rhizosphere competence of antagonistic *Fusaria* isolated from suppressive soils, Symbiosis 9, 401–404.
- Garibaldi A., Minuto A., Grasso V., Gullino M.L. (2003) Application of selected antagonistic strains against *Phytophthora cryptogea* on gerbera in closed soilless systems with disinfection by slow sand filtration, Crop Prot. 22, 1053–1061.

- Georgakopoulos D.G., Fiddaman P., Leifert C., Malathrakis N.E. (2002) Biological control of cucumber and sugar beet damping-off caused by *Pythium ultimum* with bacterial and fungal antagonists, J. Appl. Microbiol. 92, 1078–1086.
- Gold S.E., Stanghellini M.E. (1985) Effects of temperature on *Pythium* root rot of spinach grown under hydroponic conditions, Phytopathology 75, 333–337.
- Goldberg N.P., Stanghellini M.E., Rasmussen S.L. (1992) Filtration as a method for controlling *Pythium* root rot of hydroponically grown cucumbers, Plant Dis. 76, 777–779.
- Graham N., Collins R. (1996) Advances in slow sand and alternative biological filtration, Wiley, Chischester, UK.
- Grasso V., Minuto A., Garibaldi A. (2003) Selected microbial strains suppress *Phytophthora cryptogea* in gerbera crops produced in open and closed soilless systems, Phytopathol. Mediterr. 42, 55–64.
- Gruyter J., Van Kesteren H.A., Noordeloos M.E., Paternotte S.J., Veenbaas-Rijks J.W. (1992) The association of *Humicola fuscoatra* with corky root symptoms in wilted glasshouse tomatoes, Netherlands J. Plant Pathol. 98, 257–260.
- Gullino M.L., Garibaldi A. (2007) Critical aspects in management of fungal diseases of ornamental plants and directions in research, Phytopathol. Mediterr. 46, 135–149.
- Hagn A., Engel M., Kleikamp B., Munch J.C., Schloter M., Bruns C. (2008) Microbial community shifts in *Pythium ultimum*-inoculated suppressive substrates, Biol. Fertil. Soils. 44, 481–490.
- Herrero M.L., Hermansen A., Elen O.N. (2003) Occurrence of *Pythium* spp. and *Phytophthora* spp. in Norwegian greenhouses and their pathogenicity on cucumber seedlings, J. Phytopathol. 151, 36–41.
- Hoitink H.A.J., Boehm M.J. (1999) Biocontrol within the context of soil microbial communities: a substrate dependent phenomenon, Annu. Rev. Phytopathol. 37, 427–446.
- Jager G., ten Hope A., Velvis H. (1979) Hyperparasites of *Rhizoctonia solani* in Dutch potato fields, Netherlands J. Plant Pathol. 14, 86–91.
- Jenkins S.F., Averre C.W. (1983) Root diseases of vegetables in hydroponic culture systems in North Carolina greenhouses, Plant Dis. 67, 968–970.
- Jones E.E., Deacon W. (1995) Comparative physiology and behaviour of the mycoparasites *Pythium acanthophoron*, *P. oligandrum* and *P. mycoparasiticum*, Biocontrol Sci. Technol. 5, 27–39.
- Khalil S., Alsanius B. (2001) Dynamics of the indigenous microflora inhabiting the root zone and the nutrient solution of tomato in a commercial closed greenhouse system, Gartenbauwissenschaft 66, 188–198.
- Khalil S., Alsanius B., Hultberg M., Jensén P., Sundin P. (2001a) Assessment of the microbial status in closed hydroponic system using phospholipid fatty acid analysis, Acta Hort. 548, 223–227.
- Khalil S., Bååth E., Alsanius B., Englund J.E., Sundin P., Gertsson U.E., Jensén P. (2001b) A comparison of sole carbon source utilization patterns and phospholipid fatty acid profiles to detect changes in the root microflora of hydroponically grown crops, Can. J. Microbiol. 47, 302–308.
- Koohakan P., Ikeda H., Jeanaksorn T., Tojo M., Kusakari S.I., Okada K., Sato S. (2004) Evaluation of the indigenous mi-

croorganisms in soilless culture: occurrence and quantitative characteristics in the different growing systems, Scientia Hort. 101, 179–188.

- Le Floch G., Benhamou N., Mamaca E., Salerno M.I., Tirilly Y., Rey P. (2005) Characterisation of the early events in atypical tomato root colonisation by a biocontrol agent, *Pythium oligandrum*, Plant Physiol. Biochem. 43, 1–11.
- Le Floch G., Rey P., Déniel F., Benhamou N., Picard K., Tirilly Y. (2003) Enhancement of development and induction of resistance in tomato plants by the antagonist, *Pythium oligandrum*, Agronomie 23, 455–460.
- Le Floch G., Tambong J., Vallance J., Tirilly Y., Lévesque C.A., Rey P. (2007) Rhizosphere persistence of three *Pythium oligandrum* strains in tomato soilless culture assessed by DNA macroarray and real-time PCR, FEMS Microbiol. Ecol. 61, 317–326.
- Lemanceau P., Maurhofer M., Défago G. (2006) Contribution of studies on suppressive soils to the identification of bacterial biocontrol agents and to the knowledge of their modes of action, in: Gnanamanickam S.S. (Ed.), Plant-associated bacteria, Springer Netherlands, pp. 231–267.
- Lévesque C.A., De Cock A.W.A.M. (2004) Molecular phylogeny and taxonomy of the genus *Pythium*, Mycol. Res. 108, 1363–1383.
- Lifshitz R., Stanghellini M.E., Baker R. (1984) A new species of *Pythium* isolated from soil in Colorado, Mycotaxon 20, 373–379.
- Linde C., Kemp G.H., Wingfield M.J. (1994) *Pythium irregulare* associated with *Pinus* seedling death on previously cultivated lands, Plant Dis. 78, 1002–1005.
- Matta A. (1978) Fusarium tabacinum (Beyma) W. Gams, pathogen on basil and tomato in nature, Riv. Patol. Veg. 14, 119–125.
- McPherson G.M., Harriman M.R., Pattison D. (1995) The potential for spread of root diseases in recirculating hydroponic systems and their control with disinfection, Med. Fac. Landbouww. Univ. Gent. 60/2b, 371–379.
- Menzies J.A., Ehret D.L., Koch C., Bogdanoff C. (1998) *Humicola fuscoatra* infects tomato roots, but is not pathogenic, Eur. J. Plant Pathol. 104, 769–775.
- Minuto A., Clematis F., Gullino M.L., Garibaldi A. (2007) Induced suppressiveness to *Fusarium oxysporum* f. sp. *Radicis lycopersici* in rockwool substrate used in closed soilless systems, Phytoparasitica 35, 77–85.
- Minuto A., Garibaldi A., Gullino M.L. (1997) Basil an example of an approach to the protection of minor crop, Inf. Fitopatol. 47, 5–17.
- Mirza M. S., Aslam M., Ahmed Y. (1995) Sunflower wilt caused by *Fusarium tabacinum* in Pakistan, Helia 18, 91–94.
- Moorman G.W., Kang S., Geiser D.M. (2002) Identification and characterization of *Pythium* species associated with greenhouse floral crops in Pennsylvania, Plant Dis. 86, 1227–1231.
- Moulin F., Lemanceau P., Alabouvette C. (1994) Pathogenicity of *Pythium* species on cucumber in peat-sand, rockwool and hydroponics, Eur. J. Plant Pathol. 100, 3–7.
- Naseby D.C., Pascual J.A., Lynch J.M. (2000) Effect of biocontrol strains of *Trichoderma* on plant growth, *Pythium ultimum* populations, soil microbial communities and soil enzyme activities, J. Appl. Microbiol. 88, 161–169.

- Pagliaccia D., Ferrin D., Stanghellini M.E. (2007) Chemobiological suppression of root-infecting zoosporic pathogens in recirculating hydroponic systems, Plant Soil 299, 163–179.
- Pagliaccia D., Merhaut D., Colao M.C., Ruzzi M., Saccardo F., Stanghellini M.E. (2008) Selective enhancement of the fluorescent pseudomonad population after amending the recirculating nutrient solution of hydroponically grown plants with a nitrogen stabilizer, Microb. Ecol. 56, 538–554.
- Pascoe, I.G., Nancarrow R.J., Copes C.J. (1984) Fusarium tabacinum on tomato and other hosts in Australia, Trans. Br. Mycol. Soc. 82, 343–345.
- Paulitz T.C., Bélanger R.R. (2001) Biological control in greenhouse systems, Annu. Rev. Phytopathol. 39, 103–133.
- Postma J. (2004) Suppressiveness of root pathogens in closed culture systems, Acta Hort. 644, 503–510.
- Postma J., Bonants P.J.M., van Os E.A. (2001) Population dynamics of *Pythium aphanidermatum* in cucumber grown in closed systems, Med. Fac. Landbouwuniv. Gent. 66, 47–59.
- Postma J., Geraats B.P.J., Pastoor R., van Elsas J.D. (2005) Characterization of the microbial community involved in the suppression of *Pythium aphanidermatum* in cucumber grown on rockwool, Phytopathology 95, 808–818.
- Postma J., Stevens L.H., Wiegers G.L., Davelaar E., Nijhuis E.H. (2009) Biological control of *Pythium aphanidermatum* in cucumber with a combined application of *Lysobacter enzymogenes* strain 3.1T8 and chitosan, Biol. Control 48, 301–309.
- Postma J., van Os E.A., Kritzman G. (1999) Prevention of root diseases in closed soilless growing systems by microbial optimization, Acta Hort. 532, 97–102.
- Postma J., Willemsen-de Klein M.J.E.I.M., van Elsas J.D. (2000) Effect of the indigenous microflora on the development of root and crown rot caused by *Pythium aphanidermatum* in cucumber grown on rockwool, Phytopathology 90, 125–133.
- Price D. (1980) Fungal flora of tomato roots in nutrient film culture, Acta Hort. 98, 269–275.
- Rafin C. (1993) Les Pythium spp. à sporanges filamenteux, agents de nécroses racinaires sur tomate (Lycopersicon esculentum) en culture hors-sol, Thèse de doctorat, Université de Bretagne Occidentale, 166 p.
- Rafin C., Tirilly Y. (1995) Characteristics and pathogenicity of *Pythium* spp. associated with root rot of tomatoes in soilless culture in Brittany, France, Plant Pathol. 44, 779–785.
- Renault D. (2007) Caractérisation des écosystèmes microbiens colonisant les biofiltres, les solutions nutritives et les racines de *Lycopersicon esculentum* en culture hors-sol, Thèse de doctorat, Université de Bretagne Occidentale, 195 p.
- Renault D., Déniel F., Maurice S., Barbier G., Rey P. (2008) Inoculation by antagonistic bacteria of slow-filtration unit for soilless cultures : consequences on microbial communities colonizing the nutrient solutions, Phytopathology 98, S132.
- Rey P., Benhamou N., Le Floch G., Salerno M.I., Thuillier E., Tirilly Y. (2005) Different interactions between the mycoparasite *Pythium oligandrum* and two sclerotia-forming plant pathogenic fungi: *Botrytis cinerea* and *Sclerotinia minor*, Mycol. Res. 109, 779–788.
- Rey P., Benhamou N., Tirilly Y. (1998) Ultrastructural and cytochemical investigation of asymptomatic infection by *Pythium* sp., Phytopathology 88, 234–244.

- Rey P., Déniel F., Vasseur V., Benhamou N., Tirilly Y. (2001) Evolution of *Pythium* spp. populations in soilless cultures and their control by active disinfecting methods, Acta Hort. 554, 341–348.
- Rey P., Le Floch G., Benhamou N., Tirilly Y. (2008) *Pythium oligandrum* biocontrol: its relationships with fungi and plants, in: Ait Barka E., Clément C. (Eds.), Plant-microbe interactions, pp. 43–57.
- Rey P., Leucart S., Desilets H., Belanger R., Larue J.P., Tirilly Y. (2001) Production of auxin and tryptophol by *Pythium ultimum* and minor pathogen, *Pythium* group F. Possible role in pathogenesis, Eur. J. Plant Pathol. 107, 895–904.
- Rey P., Nodet P., Tirilly Y. (1997) *Pythium F* induces a minor but ubiquitous disease in tomato soiless cultures, J. Plant Pathol. 79, 173–180.
- Rey P., Picard K., Déniel F., Benhamou N., Tirilly Y. (1999) Development of an IPM system in soilless culture by using slow sand filtration and a biocontrol agent, *Pythium oligandrum*, in: van Leuteren J.C. (Ed.), Integrated control in glasshouses, IOBC wprs Bulletin 22, 205–208.
- Runia W.T. (1995) A review of possibilities for disinfection of recirculation water from soilless cultures, Acta Hort. 382, 221–229.
- Soran H., Ozel M. (1985) Light and electron microscopic investigation on roots of Cucumis melon plants inoculated with *Fusarium* species, J. Turkish Phytopathol. 14, 69–77.
- Spadaro D., Gullino M.L. (2005) Improving the efficacy of biocontrol agents against soilborne pathogens, Crop Prot. 24, 601–613.
- Stanghellini M.E., Kronland W. (1986) Yield loss in hydroponically grown lettuce attributed to subclinical infection of feeder rootlets by *Pythium dissotocum*, Plant Dis. 70, 1053–1056.
- Stanghellini M.E., Rasmussen S.L. (1994) Hydroponics a solution for zoosporic pathogens, Plant Dis. 78, 1129–1138.
- Steinberg C., Moulin F., Gaillard P., Gautheron N., Stawiecki K., Bremeersch P., Alabouvette C. (1994) Disinfection of drain water in greenhouses using a wet condensation water, Agronomie 14, 627–635.
- Tambong J.T., De Cock A.W.A.M., Tinker N.A., Lévesque C.A. (2006) An oligonucleotide array for identification and detection of *Pythium* species, Appl. Environ. Microbiol. 72, 2691–2706.
- Termorshuizen A.J., van Rijn E., van der Gaag D.J., Alabouvette C., Chen Y., Lagerlöl J., Malandrakis A.A., Paplomatas E.J., Rämert B., Ryckeboer J., Steinberg C., Zmora-Nahum S. (2006) Suppressiveness of 18 composts against 7 pathosystems : variability in pathogen response, Soil Biol. Biochem. 38, 2461–2477.
- Tirilly Y., Letard L. (1997) Maîtrise sanitaire des solutions nutritives en cultures hors sans sol, Infos-Ctifl. 132, 35–39.
- Tu J.C., Papadopoulos A.P., Hao X., Zheng J. (1999) The relationship of *Pythium* root rot and rhizosphere microorganisms in a closed circulating and an open system in rockwool culture of tomato, Acta Hort. 481, 577–583.
- Vallance J., Le Floch G., Déniel F., Barbier G., Lévesque C.A., Rey P. (2009) *Pythium oligandrum* biocontrol in the

rhizosphere: influence on fungal and oomycete population dynamics, Appl. Env. Microbiol. 75, 4790–4800.

- van der Gaag D.J., Wever G. (2005) Conduciveness of different soilless growing media to *Pythium* root and crown rot of cucumber under near-commercial conditions, Eur. J. Plant Pathol. 112, 31–41.
- van Os E.A., Amsing J.J., van Kuik A.J., Willers H. (1999) Slow sand filtration: a potential method for the elimination of pathogens and nematodes in recirculating nutrient solutions from glasshouse-grown crops, Acta Hort. 481, 519–526.
- van Os E.A., Bruins M., Postma J., Willemsen-de Klein M.J.E.I.M. (2004) Investigations on crop developments and microbial suppressiveness of *Pythium aphanidermatum* after disinfection treatments of the circulating nutrient solution, Acta Hort. 644, 563–570.
- van der Plaats-Niterink J. A. (1981) Monograph of the genus *Pythium*, Studies in Mycology 21, 1–242.
- Vasseur V., Rey P., Bellanger E., Brygoo Y., Tirilly Y. (2005) Molecular characterization of *Pythium* group F isolates by ribosomal- and intermicrosatellite-DNA regions analysis, Eur. J. Plant Pathol. 112, 301–310.
- Waechter-Kristensen B., Gertsson U.E., Sundin P. (1994) Prospects for microbial stabilization in the hydroponic culture of tomato using circulating nutrient solution, Acta Hort. 361, 382–387.
- Waechter-Kristensen B., Khalil S., Sundin P., Englund J.E., Gertsson U.E., Jensén P. (1996) Study of the microbial dynamics in the root environment of closed, hydroponic cultivation systems for tomato using phospholipid fatty acid profiles, Acta Hort. 440, 193–198.
- Waechter-Kristensen B., Sundin P., Gertsson U.E., Hultberg M., Khalil S., Jensén P., Berkelmann-Loehnertz B., Wohanka W. (1997) Management of microbial factors in the rhizosphere and nutrient solution of hydroponically grown tomato, Acta Hort. 450, 335–339.
- Weber-Shirk M.L., Dirk R.I. (1997) Physical-chemical mechanisms in slow sand filters, Am. Water Works Assoc. J. 89, 87–100.
- Weller D.M., Raaijmakers J.M., MsSpadden Gardener B.B., Thomashow L.S. (2002) Microbial populations responsible for specific soil suppressiveness to plant pathogens, Annu. Rev. Phytopathol. 40, 309–348.
- Whipps J.M., Lumsden R.D. (1991) Biological control of *Pythium* species, Biocontrol Sci. Technol. 1, 75–90.
- Wohanka W., Luedtke H., Ahlers H., Luebke M. (1999) Optimization of slow filtration as a mean for disinfecting nutrient solutions, Acta Hort. 481, 539–544.
- Wulff E.G., Pham A.T.H., Chérif M., Rey P., Tirilly Y., Hockenhull J. (1998). Inoculation of cucumber roots with zoospores of mycoparasitic and plant pathogenic *Pythium* species: differential zoospore accumulation, colonization ability and plant growth response, Eur. J. Plant Pathol. 104, 69–76.
- Zhang W., Tu J.C. (2000) Effect of ultraviolet disinfection of hydroponic solutions on *Pythium* root rot and non-target bacteria, Eur. J. Plant Pathol. 106, 415–421.

## **Allelopathy in Compositae Plants**

S.-U. Chon and C.J. Nelson

#### Contents

1	Intro	luction	728
2	Allelo	pathic Compositae Plants	729
	2.1	Artemisia sp	729
	2.2	Cirsium sp	730
	2.3	Lactuca sp	731
	2.4	Xanthiun sp	731
3	From	Petri-Dish Bioassays to Field Trials	731
	3.1	Bioassays in Petri-Dish	731
	3.2	Morphological Responses	732
	3.3	Residue Incorporation in Soil	733
	3.4	Field Experiments	733
4	Disco	very of Causative Allelochemicals	734
	4.1	Compounds Involved	734
	4.2	Identification and Quantification	734
	4.3	Fractionation and Bioassay of Each Solvent	
		Fractions	735
	4.4	Phytotoxicity of Major Causative Allelopathic	
		Compounds	735
Re	ference	es	736

S.-U. Chon (🖂)

tural management such as weed control, crop protection, and crop re-establishment. Compositae plants have potent allelopathic activity, and the activity is confirmed through (a) bioassays with aqueous or various solvent extracts and residues, (b) fractionation, identification, and quantification of causative allelochemicals, and (c) mechanism studies on the allelochemicals. Most assessments of allelopathy involve bioassays of plant or soil extracts, leachates, fractions, and residues based on seed germination and seedling growth in laboratory and greenhouse experiments. Plant growth may be stimulated below the allelopathic threshold, but severe growth reductions may be observed above the threshold concentration depending upon the sensitivity of the receiving species. Generally germination is less sensitive than is seedling growth, especially root growth. Some approaches showed that field soil collected under donor plants significantly reduced or somewhat promoted growth of the test plants. Petri-dish bioassays with methanol extracts or fractions and causative phenolic allelochemicals showed significant phytotoxic activities in concentrationdependent manner. Delayed seed germination and slow root growth due to the extracts could be confounded with osmotic effects on rate of imbibition, delayed initiation of germination, and especially cell elongation; the main factor that affects root growth before and after the tip penetrates the seed coat. Light and electron microscopic approaches extract evaluation at the ultrastructural level have been precisely investigated. Many Compositae plants have allelopathic potentials, and the activities and types and amount of causative compounds differ depending on the plant species. The incorporation of allelopathic substances

Abstract Allelopathy plays an major role in agricul-

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_32,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Chon and Nelson, Agron. Sustain. Dev. 30 (2010) 349–358. DOI: 10.1051/agro:2009027

EFARINET Co. Ltd., BI Center, Chosun University, Gwangju 501-759, South Korea e-mail: chonsu4100@yahoo.co.kr

into agricultural management may reduce the use of pesticides and lessen environmental deterioration.

**Keywords** Allelopathy • Compositae plants • Eco-friendly weed control • Bioassay • Extracts • Fractions • Residues • Water-soluble • Activation mechanism • Phenol compounds

#### 1 Introduction

Allelopathy was defined by Molisch (1937) as a chemical interaction between plants or sometimes between microbes and higher plants that includes stimulatory as well as inhibitory influences. Later it was defined as any direct or indirect, harmful or beneficial effect of one plant as a donor plant on another as a recipient plant through the production of chemical compounds that escape into the environment (Rice, 1984). When a plant produces allelopathic compounds that are detrimental to establishment of new seedlings of the same species, the effect is called autotoxicity which is a specialized intraspecific form of allelopathy. Allelopathy and autotoxicity can play significant roles under both natural and manipulated ecosystems (Rice, 1984), mainly by adversely affecting seed germination and seedling growth. Allelopathic interactions among plants have been observed for centuries even though few specific allelochemicals have been identified.

Allelopathy plays a key role in weed control, crop protection, and crop re-establishment. Suitable manipulation of allelopathy towards improvement of crop productivity and environmental protection through eco-friendly control of weeds, pests, crop diseases, conservation of nitrogen in crop land, and synthesis of novel agrochemicals based on natural products have gained prominent attention of scientists engaged in allelopathic research. The allelochemicals can affect physiological functions like respiration, photosynthesis and ion uptake. More recently, however, scientific attention has also been drawn to exploit the positive significant roles of allelopathy and what role this phenomenon can play in enhancing crop productivity.

A serious problem of modern agriculture is crop loss caused by weeds. Worldwide, weeds alone cause a 10% loss of agriculture production (Altieri and Liebman, 1988). Yet, allelopathic principles of crops can be used as an alternative mean of weed control based on natural products. Although allelopathy is often considered a problem for agriculture, there is now considerable evidence to suggest that it might be exploited to help manage weed problems in a variety of agroecosystems. In agroecosystems, several weeds, crops, agroforestry trees and fruit trees have been shown to exert allelopathic influence on associated or subsequent crops, thus, affecting their germination and growth adversely (Kohli et al., 1993). Most allelopathic evidence has been associated with the effect of weeds on crops and crops on crops, and crops on weeds. Of these, an important economical potential of allelopathy may be the ability of crops to suppress weeds. Allelopathic weeds also can affect crops by a number of ways like delaying or preventing seed germination and reducing seedling growth. Approaches that have already been explored are selection for allelopathic types within the germplasm of crops, use of allelopathic rotational or companion plants in cropping systems, and biosynthesis of useful natural herbicides from higher plants and microorganisms.

Alternatives to synthetic chemical herbicides need to be developed, especially for organic or eco-friendly farming operations, landscape management systems, home gardens, and for situations where public policies mandate reduced pesticide use. Most studies on allelopathy have focused on interference and allelopathic effects of several important weeds on crop yields. Several weeds have been shown to have allelopathic potentials and others are suspected to have allelopathic potential in agro-ecosystems (Rice, 1984). Plant seedlings of various crops possess allelopathic potential or weed-suppressing activity, including cucumber (Cucumber sativs L.) (Putnam and Duke, 1974), oat (Avena spp.) (Fay and Duke, 1977) and rice (Oryza sativa L.) (Dilday et al., 1994; Olofsdotter and Navarez, 1996). A total of 538 accessions of cultivated and wild cucumber were screened in pot and field tests with several accessions causing inhibited growth of Brassica hirta and Panicum miliaceum (Putnam and Duke, 1974). Out of more than 3000 accessions of oat, several exuded a large amount of an allelochemical, scopoletin (Fay and Duke, 1977). Also oat suppressed the growth of Erysimum cheiranthoides in both laboratory and field tests due at least in part to an allelopathic mechanism (Markova, 1972).

Improving the competitive ability of crops also reduces dependency on herbicides. However, attempts to increase competitive ability while maintaining productivity have had limited success and no crop cultivar has been released with superior competing ability as a marketing argument. In crop competition including allelopathy, the importance of chemical interference has often been discussed (Rice, 1995). The incorporation of an allelopathic character into a crop cultivar could provide the plant with a means of gaining a competitive advantage over certain important weeds (Putnam and Duke, 1974). Wu et al. (1999) suggested that genetically improving crops with allelopathic potential and the allelopathy can play an important role in future weed management.

#### 2 Allelopathic Compositae Plants

Twelve weed families include 68% of the 200 species that are the most important world weeds. Of them, Compositae plants account for 16% of the world's worst weeds are included (Holm, 1978). Allelopathy in weeds has been found in a number of plants employing various laboratory assays. Some of previous studies on allelopathic effects of weeds have been several Compositae plant species such as *Artemisia*, *Cirsium*, *Lantuca* and *Xanthium* species.

Inam et al. (1987) reported that aqueous extracts of Xanthium strumarium from different plant parts reduce germination, early growth and dry weight of Brassica compestris, Lactuca sativa, and Pennisetum americanum. Parthenium hysterophours is also known to be very allelopathic to wheat (Kanchan and Jayachandra, 1979), soybean, and corn (Mersie and Singh, 1978). Extracts and residues of these plants significantly reduced germination and root and shoot dry weight of the test plants. Bendall (1975) studied water and ethanol extracts and residues in soil and concluded that an allelopathic mechanism might be involved in the exclusion of some annual thistle (Carduus crispus L.), pasture, and crop species in areas infested with Cirsium arvense (L.) Scop. C. arvense litter reduced the growth of Amaranthus retroflexus L. and Setaria viridis L. more than that of cucumber (Cucumis sativus L.) or barley (Hordeum vulgare L.) in their greenhouse experiments (Stachon and Zimdal, 1980). In a field experiment, high densities of C. arvense reduced the incidence of annual weeds growing in the same vicinity of C. arvense (Stachon and Zimdal, 1980). More

recently, Chon et al. (2003a) reported that aqueous leaf extracts from 16 Compositae plant species were bioassayed against alfalfa (*Medicago sativa*) to evaluate their allelopathic effects, and the results show the highest inhibition in the extracts from *C. japonicum*, *L. sativa*, and *X. occidentale*, all showing significant inhibition of seed germination and growth of alfalfa.

Einhellig (1986) noted that biological activity of allelochemicals, including the autotoxic factors in alfalfa, was concentration dependent with a response threshold. Plant growth may be stimulated below the threshold, with mild to severe growth reductions observed above the threshold; each depending on the sensitivity of the receiving species, the plant process affected and the existing environmental conditions. Many instances of stimulation effects of microorganisms on other organisms and of plants on microorganisms have been reported. Rice (1986) demonstrated growth stimulatory effects of volatile compounds, decaying leaves and root exudates of parasitic and nonparasitic plants on several other species. He showed that decaying ground ivy leaves stimulated seedling growth of downy brome and radish. Chon et al. (2003a) evaluted aqueous extracts of 16 Compositae plants and found Lactuca sativa, Xanthium occidentale and Cirsium japonicum showed the highest inhibition on alfalfa (Medicago sativa) seedlings. Conversely, extracts of Chrysanthemum indicum, Youngia sonchifolia, Bidens frondosa, and Breea segeta at concentrations below 20 g dry matter  $L^{-1}$  increased root length of alfalfa by 13-33%. In another study, alfalfa root growth was stimulated by very low concentrations from alfalfa leaf extracts (Chon et al., 2000). Our findings, however, indicated that this stimulatory effect for a reputed autotoxic chemical was less than those reported by Einhellig (1986) and Rice (1986) who evaluated stimulatory effects of allelochemicals. Chon et al. (2003a) reported that stimulatory as well as inhibitory effects of Compositae plant species were exhibited (Table 1).

#### 2.1 Artemisia sp.

The genus *Artemisia* comprises over 400 species of perennial herbs and shrubs which are widely distributed throughout Europe, Asia, North America and South Africa (Stairs, 1986) and the plants are

	Extract concentration, g $L^{-1}$						
Plant species	10	20	30	40			
Bidens frondosa	44.3 (128)	39.7 (115)	31.4 (91)	7.1 (20)			
Breea segetum	42.1 (122)	43.7 (126)	28.2 (81)	8.8 (25)			
Chrysanthemum indicum	41.2 (119)	45.9 (133)	35.3 (102)	26.5 (76)			
Youngia sonchifolia	43.5 (126)	40.1 (116)	36.5 (105)	27.4 (79)			
Eclipta prostrata	28.9 (84)	19.5 (56)	4.1 (12)	1.7 (5)			
Cirsium japonicum	31.5 (91)	22.0 (63)	7.6 (22)	0.0 (0)			
Xanthium occidentale	33.3 (96)	2.9 (8)	0.0 (0)	0.0 (0)			
Lactuca sativa	34.3 (99)	0.2 (1)	0.0 (0)	0.0 (0)			

 Table 1
 Effects of aqueous extracts from several *Compositae* plants on alfalfa root length (mm) 6 days after seeding. Root length of untreated control was 34.6 mm. Adapted from Chon et al. (2003a)

\* Values in parentheses represent % of control.

considered weeds in many part of the world. Many weeds are ecologically important and contain bioactive compounds such as allelopathic and antifungal constituents in order to survive in ecosystem (Meepagala et al., 2003). Allelopathic constituents are cause of the symptom of the non-growth or growth retardation of neighbouring plant. The phenomena have been found in the ethereal oils and alkaloid absinthium excreted from Artemisia absinthium (Funke, 1943), natural and artificial rain drip from Artemisia californica (Halligan, 1976), aqueous extract and volatile substances of Artemisia princeps var. orientalis (Yun, 1991) and aqueous extract of Artemisia campestris ssp. Caudate (Yun and Maun, 1997). And the genus Artemisia is rich sources of biologically active natural products (Tan et al., 1998). Approximately 30 Artemisia species grown in Korea have been used in traditional Korean medicine. Of them Artemisia princeps var. orientalis is widly used as medicinal herb and is known as a strong weed, it is a Korean custom to uproot the plant in old tomb area because its watersoluble allelochemicals retard turf growth.

Kil and Yun (1992) reported that aqueous extracts from mature leaf, stem, and root of *Artemisia princeps* var. *orientalis* caused significant inhibition in germination and decreased seedling elongation of receptor plants, whereas germination of some species was not inhibited by extracts of stem and root. Dry weight growth was slightly increased at lower concentrations of the extract, whereas it was proportionally inhibited at higher concentrations. Kil et al. (1992) also reported, an in vitro study, that aqueous extract of *A. princeps* var. *orientalis* caused some reduction in concentration, induction, and growth of callus, although they looked normal, whereas the explants of most receptor plants did not develop callus at higher concentration. They also found that lettuce and *Eclipta prostrata* were the most sensitive species, and even A. *princeps* var. *orientalis* was affected by its own extracts. Yun and Kil (1992) reported differential phytotoxicity of residues of *A. princeps* var. *orientalis* using various plants as test species in their field and laboratory studies. In seedling growth tests with abandoned field soils (control) and soil underneath the *Artemisia* plant, the elongation and dry weight of seedlings grown in the soil from under the *Artemisia* plants were severely inhibited, thereby suggesting that certain growth inhibitors were released from the *Artemisia* plant and the inhibitor remained in the soil.

#### 2.2 Cirsium sp.

Common thistle (Cirsium japonicum) is a noxious perennial weed which causes serious yield losses in spring sown small grains row crops, and pastures (Hodgson, 1963, 1968). Growth and germination of wheat (Triticum aestivum L.) and flax (Linum usitatissimum L.) were inhibited by aqueous extracts of Canada thistle (Cirsium arvense Scop.) roots and shoots (Helgeson and Konzak, 1950). Bendall (1975) studied water and ethanol extracts and residues in soil, and concluded that an allelopathic mechanism might be involved in the exclusion of some annual thistle, pasture, and crop species from Canada thistle areas. Stachon and Zimdal (1980) in their greenhouse experiments found that Canada thistle litter reduced the growth of redroot pigweed (Amaranthus retroflexus L.) and green foxtail (Setaria viridis L.) more than that of cucumber (*Cucumis sativus* L.) or barley (*Hordeum vulgare* L.). Chon et al. (2003a) reported that the aqueous leaf extracts of 16 Compositae plant species, including common thistle, significantly inhibited hypocotyls and root lengths of alfalfa. In their earlier study, *trans*-cinnamic acid was found as the greatest amount at ethyl acetate fraction from methanol extracts of common thistle plant.

Chon (2004a) assayed phytotoxic effects of a series of aqueous extracts from leaves, stems, roots and flowers of common thistle (*Cirsium pendulum* Fisch.) against alfalfa (*Medicago sativa*) seedlings. The results showed highest inhibition in the extracts from flowers and leaves, and followed by stems, and roots. He found also that hexane and ethylacetate fractions of common thistle reduced alfalfa root growth more than did butanol and water fractions. Incorporation into soil with the leaf residues at 100 g kg<sup>-1</sup> inhibited shoot fresh weights of barnyardgrass and eclipta (*Eclipta prostrate*) by 88 and 58%, respectively, showing higher sensitivity in grass species.

#### 2.3 Lactuca sp.

Lettuce (*Lactuca sativa*) is an annual herbaceous plant of Compositae, one of the largest and most diverse families of flowering plants. Major weeds including barnyard grass (*Echinochloa colonum*), common purslane (*Portulaca oleracea*), smooth pigweed (*Amaranthus hybridus*), shepherd's purse (*Capsella bursa-pastoris*) and common lambsquaters (*Chenopodium album*) are known to interfere with lettuce (Haar and Fennimore, 2003; Santos et al., 2003; Fennimore and Umeda, 2003).

Chon et al. (2005) showed phytotoxic effects of aqueous or methanol extracts and residues from leaves of lettuce cultivars on alfalfa seed germination. Methanol extracts from hexane fraction of lettuce plants showed the most inhibition on alfalfa root growth and followed by ethylacetate, butanol and water fractions. Incorporation with leaf residues of 100 g kg<sup>-1</sup> into soil significantly inhibited shoot- and root fresh weights of barnyard grass by 79 and 88%, respectively. Also, Chon et al. (2005) reported major allelopathic substances by means of HPLC were identified as coumarin, *trans*-cinnamic acid, *o*-coumaric

acid, *p*-coumaric acid and chlorogenic acid. Of them *p*-coumaric acid was found as the greatest amount (8.9 mg 100 g<sup>-1</sup>) in the EtOAc fraction. Hexane and EtOAc fractions of lettuce reduced alfalfa root growth more than did BuOH and water fractions.

#### 2.4 Xanthiun sp.

Xanthium species is one of the most competitive weeds in crop fields as well as wastelands. Inam et al. (1987) found that aqueous extracts of Xanthium strumarium from different plant parts reduce germination, early growth and dry weight of Brassica compestris, Lactuca sativa, and Pennisetum americanum. Especially, Parthenium hysterophours is known to be very allelopathic to wheat (Kanchan and Jayachandra, 1979), soyabean, and corn (Mersie and Singh, 1978). Their extracts and residues significantly reduced germination and shoot and root dry weight of the test plants.

Chon (2004b) reported that aqueous extracts from *X. occidentale* completely inhibited the hypocotyl and root growth of alfalfa. Early seedling growth of both alfalfa barley (*Hordeum vulgare* L.), soybean (*Glycine max* L.), and barnyard grass (*Echinochloa crus-galli*) was significantly reduced by methanol extracts. By means of high-performance liquid chromatography, chlorogenic acid and *trans*-cinnamic acid were quantified as the highest amounts from water and ethylacetate fractions, respectively. Butanol and ethylacetate fractions of *X. occidentale* reduced alfalfa root growth more than did hexane and water fractions.

### 3 From Petri-Dish Bioassays to Field Trials

#### 3.1 Bioassays in Petri-Dish

Most assessments of allelopathy, especially in early stages, involve bioassays of plant or soil extracts based on seed germination and seedling growth. A reliable bioassay that is sensitive is needed for more in-depth studies of the growth mechanisms involved and for developing initial analytical procedures to determine the chemical (s) responsible. Generally germination is less sensitive to the allelopathic chemical than is seedling growth, especially root growth (Miller, 1996). Although many laboratory bioassays have proposed to demonstrate allelopathy, concerns have been raised that most of them have little relevance in terms of explaining behavior in the field (Connell, 1990; Inderjit and Olofsdotter, 1998; Inderjit and Dakshini, 1995, 1999).

It is a complex phenomenon involving a variety of inter-relationships among plants. Virtually all plant parts such as leaves (Kumari and Kohli, 1987), roots (Horsley, 1977), pollen (Cruz-Ortega et al., 1988), trichomes (Bansal, 1990), bark (Kohli, 1990) and seeds and fruits (Fredman et al., 1982) have allelopathic potential. It is generally accepted that water extracts of top growth (especially leaves) produce more allelopathy for seedlings than those from roots and crowns of alfalfa (Medicago sativa L.) (Miller, 1996), and that shoot extract from the reproductive stage was more inhibitory than from the vegetative stage under laboratory conditions (Chung and Miller, 1995a; Hegde and Miller, 1992a). Chung and Miller (1995a) ranked autotoxic effects of water extracts of plant parts of alfalfa as leaf (greatest), seed, root, flower, and stem (least). Chou and Leu (1992) reported that extracts from flowers of Delonix regia (BOJ) RAF exhibited highest inhibition against three test plants, alfalfa, lettuce (Lactuca sativa), and Chinese cabbage (Brassica chinensis). Chon (2004a) reported that phytotoxic effects of a series of aqueous extracts from leaves, stems, roots and flowers of common thistle (Cirsium pendulum Fisch.) on alfalfa (Medicago sativa) seedlings showed highest inhibition in the extracts from flowers and leaves, and followed by stems, and roots (Fig. 1).

#### 3.2 Morphological Responses

Some plant genotypes are likely to escape the allelopathic chemical (s) by being hypersensitive. In this regard the root tip may actually be strongly affected by allelochemical (s) and have its growth rate nearly stopped. But if the seedling quickly enables to produce several secondary roots, the number of apices per soil volume increases at higher position in soil profile. A study demonstrated microscopically that alfalfa extract reduced both root growth and root hair density of alfalfa (Hegde and Miller, 1992b). Stimulation of



**Fig. 1** Effects of *Cirsium japonicum* leaf, stem, root and flower extracts on shoot (**a**) and root lengths (**b**) of alfalfa as affected by different concentrations. Within an extract concentration, means followed by the same letter are not significantly different at P < 0.05. Each bar represents standard error of the mean. Adapted from Chon et al. (2003a)

lateral root growth to the detriment of the primary root also suggests disruption of hormonal balance (Dayan et al., 2000). Anatomical responses of tissue cells upon water-soluble substances or allelochemicals need to be elucidated. The morphology of seedlings grown in the presence of a phytotoxin may also yield important information. Stimulation of lateral root growth to the detriment of the primary root also suggests disruption of hormonal balance (Dayan et al., 2000). Benzoic and cinnamic acids-treated soybean plants showed fewer lateral roots and tended to grow more horizontally compared to the untreated plants. Their lateral roots were stunted and less flexible (Baziramakenga et al., 1994).

Not many microscopic approaches at ultrastructural level have been conducted on allelopathic effects of extracts or allelochemicals. Chon et al. (2002) suggested that the root systems, especially root tips of alfalfa, were stunted and swollen by the aqueous alfalfa leaf extracts at 30 g  $L^{-1}$  and coumarin at  $10^{-3}$  M. Duke et al. (1987) discovered artemisinin, a constituent of annual wormwood (Artemisia annua), marginally increased the mitotic index of lettuce root tips at 33 µM. At the ultrastructural level by means of electron microscopic study, however, chromosomes were less condensed during mitosis in artemisinin-treated than control meristematic cells. Liu and Lovett (1993) demonstrated that barley allelochemicals, hordenine and gramine affected damage of cell walls, increase in both size and number of vacuoles, autophagy, and disorganization of organelles. More recently, a study on allelopathic interference of benzoic acid against mustard (Brassica juncea L.) seedling growth showed irregular shaped cells arranged in a disorganized manner and disruption of cell organelles at cellular level (Kaur et al., 2005). Their result indicates that damage to the mustard root at cellular level was clearly evidenced by the changes in cell morphology and internal organization.

#### 3.3 Residue Incorporation in Soil

Generally, residue inhibition of seedling growth was enhanced if crop residue was incorporated before planting, drastically reduced if residue remained on the surface (Cochran et al., 1980; Elliott et al., 1981). Crop residue toxicity to winter wheat seedlings was likely caused by either an allelopathic compound or N immobilization due to increased microbial populations. The allelopathic compound was either a watersoluble compound leached from residue or a compound produced during microbial decomposition of plant residue (Cochran et al., 1980; Elliott et al., 1981). Another study (Kadioglu, 2000) showed that the inhibitory effect of ground hearleaf cocklebur (Xanthium strumarium L.) on Amaranthus retroflexus, Amaranthus sterilis, and Conium maculatum. Chon and Kim (2005) reported that the effect of residue incorporation with Xanthium occidentale plant samples into soil on seedling growth of barnyard grass was examined in the greenhouse, and results showed that the leaf residues at 100  $g kg^{-1}$  inhibited shoot and root dry weights of test plants by 70-90% (Fig. 2).



**Fig. 2** Effects of incorporation with dried leaf material of *Xanthium occidentale, Lactuca sativa,* and *Cirsium japonicum* on shoot (**a**) and root (**b**) fresh weight of barnyard grass 15 days after seeding or treatment. Material was incorporated by mixing with soils. After incorporation, barnyardgrass was seeded and immediately subirrigated. Shoot and root fresh fresh weights were measured 15 days later. Adapted from Chon and Kim (2005)

#### 3.4 Field Experiments

A field research described that the early stage of grassland succession in Korea would be composed of *Plantago asiatica*, *Artemisia princeps* var. *orientalis*, *Oenothera odorata*, and *Zoysia japonica* was inhibited greatly by the *Artemisia* extracts. These contrasting views could result from differences between laboratory work and field observations (Park, 1966). Yun and Kil (1992) reported that field soil collected under the *Artemisia* plants significantly reduced or somewhat promoted growth of the test plants. These results are in agreement or disagreement with following results: soil collected from some plant fields exhibited phytotoxicity by reducing growth (Al-Naib and Rice, 1971;

AlSaadawi and AlRubeaa, 1985; Inam et al., 1989) and soil from below some test plants did not inhibit germination and growth of the test species (Fadayomi and Oyebade, 1984; Goel and Sareen, 1986).

## 4 Discovery of Causative Allelochemicals

#### 4.1 Compounds Involved

Most important approach on allelopathy is to successfully isolate, identify, and quantify causative allelochemicals that present in plants or soils. It is essential that potential allelopathic compounds can also relate to the levels originally in the whole extracts. Natural products identified as allelopathic agents have been classed into the following (a) toxic gases, (b) organic acids from Krebs cycle and aldehydes, (c) aromatic acids, (d) simple unsaturated lactones, (e) courmarins, (f) quinones, (g) flavonoids, (h) tannins, (i) alkaloids, and (j) terpenoids and steroids, etc. Although many of these compounds are secondary products of plant metabolism, several are degradation products that occur in the presence of microbial enzymes. Several biosynthetic pathways lead to production of the various categories of allelopathic agents. The inhibitors usually arise through either the acetate or through the shikimic acid pathway. Several types of inhibitors, which originated from amino acids, come through the acetate pathway. Most of compounds that cause allelopathy were derived from amino acids, via the shikimic acid pathway (Rice, 1984).

The source of the active agents may be living plants, litter, detritus, leachates, soil bateria and fungi, mycorhizal fungi, root exudates, the atmosphere, water, air-borne particles, or pathogenic organisms. Many organisms may be involved simultaneously in a particular interaction. Compounds isolated from plants or their leachates often do not reproduce the observed allelopathic effects without the associated factors. In many instances, these plant-derived compounds are modified by oxidation, reduction, photochemical activation (Fisher et al., 1994), detoxification, or biochemical activation by bacteria and fungi. Bioactive molecules may leave the system by being adsorved onto inorganic particles or organic matter in the soil, or leached from the system.

#### 4.2 Identification and Quantification

Causative allelochemicals reported from Compositae plant species have not been identifieed sufficient. Identifying and quantifying the causative allelochemicals in plan and associated environments are the most important approaches for allelopathy study. Phenolic acids in the literature on allelopathy are often mentioned as putative allelochemicals and are perhaps the most commonly investigated compounds among potential allelochemicals. They are found in a wide range of soils or plants, and their phytotoxic potential against various plants has been demonstrated under controlled conditions. Phytotoxicity-based extraction and fractionation were employed to separate allelochemicals contained in each plant extracts. Chon et al. (2003a) reported that by means of high-performance liquid chromatography (HPLC) analysis, the responsible causative allelopathic substances present in L. sativa, X. occidentale, and C. japonicum were isolated from various fractions and identified as coumarin, trans-cinnamic acid, ocoumaric acid, p-coumaric acid, and chlorogenic acid.

**Table 2** Quantitative determination using HPLC on concentration of some phenolic compounds present in leaves of *Xanthium occidentale* (Chon et al., 2003a)

	Fractions							
Compound	Hexane	EtOAc	BuOH	Water	Total			
			- mg 100 g <sup>-1</sup> -					
Coumarin	0.14	2.07	2.02	0.19	4.41			
trans-cinnamic acid	8.39	20.16	4.20	0.2	32.95			
p-coumaric acid	nd	0.24	nd	nd	nd			
Chlorogenic acid	2.24	0.24	3.59	33.33	39.40			
Total	10.77	22.67	9.81	33.54	76.76			

S.-U. Chon and C.J. Nelson

<sup>a</sup> nd: none detected.

## 4.3 Fractionation and Bioassay of Each Solvent Fractions

Methanol extracts from ground plant samples are used for the following bioassay and fractionation. Generally, for instance, crude methanol extracts are diluted with distilled water and hexane to collect hexane fraction using a separating funnel. After hexane collection, the distilled water fractions are added with ethylacetate to obtain ethylacetate fraction in the same way. The same procedure is used in preparing other solevent fractions. The fractions are taken to dryness on a rotary evaporator at 40-50°C, and transferred into vacuum freeze dryer to obtain dry matters and used for quantification and bioassay. The dried samples concentrated from fractions are again dissolved in MeOH to use for bioassay (Chon et al., 2003a). Each of these fraction solutions are pippeted in a 9-cm plastic Petri dish lined with one Whatman No. 1 filter paper and evaporated to dryness for 24 h at 24°C. For the control, 4 mL of methanol is applied to Petri dishes. After evaporation, distilled water is added onto the filter paper and then seeds of test plant are placed on the paper and grown for 6 days. Chon et al. (2003a) reported that these major phenolic compounds present in Compositae species were total phenol compounds of all fractions in C. japonicum, L. sativa, and X. occiden*tale* by 60.3, 18.5, and 84.4 mg 100  $g^{-1}$ , respectively. They reported also, through the bioassay procedure, X. occidentale which had the highest total concentration, showed the most inhibitory effect on test plants in Compositae plant species (Table 2).

## 4.4 Phytotoxicity of Major Causative Allelopathic Compounds

Chon et al. (2003a) reported that among 10 phenolic compounds assayed for their phytotoxicity on root growth of alfalfa, coumarin, *trans*-cinnamic acid and *o*-coumaric acid were most inhibitory. Especially, coumarin at  $10^{-3}$  M significantly inhibited root growth of alfalfa. Methanol extracts from BuOH, EtOAc, hexane, and water fractions were also assayed to confirm their phytotoxic effects. The results showed that methanol extracts of *X. occidentale* were most inhibitory on root growth of alfalfa, and that



**Fig. 3** Effects of various fractions from methanol extracts of *Cirsium japonicum* (**a**), *Lactuca sativa* (**b**), and *Xanthium occidentale* (**c**) on alfalfa root length 6 days after seeding. Adapted from Chon et al. (2002)

methanol extracts from BuOH and EtOAc fractions of *X. occidentale* reduced alfalfa root growth more than did those from hexane and water fractions. Methanol extracts from hexane and EtOAc fractions of *L. sativa* and *C. japonicum* reduced alfalfa root growth more than did those of BuOH and water fractions. Especially, methanol extracts from hexane and EtOAc

fractions at 50 g L<sup>-1</sup> reduced root growth by each 85%, while treatment at same concentration of BuOH and water fractions reduced root growth by 40 and 15%, respectively (Fig. 3).

Acknowledgements This research was conducted with support from a part of the 2005 MAF-ARPC research fund (105088-33-1-HD110). Appreciation is expressed to Dr. Young-Min Kim at Biotechnology Industrialization Center, Dongshin University, Naju, Korea, for his technical assistance.

#### References

- Al-Naib F.A., Rice E.L. (1971) Alleolpathic effect of *Platanus* occidentalis, Bull. Torrey Botancal Club 98, 75–82.
- AlSaadawi I.S., AlRubeaa A.J. (1985) Allelopathic effects of *Citrus aurantium* L. I. Vegetational patternig, J. Chem. Ecol. 11, 1515–1525.
- Altieri A.M., Liebman M. (1988) Weed management in agroecosystem: Ecological approaches, Florida: CRC Press.
- An M., Pratley J.E., Haig T. (2001) Phytotoxicity of vulpia residues: III. Biological activity of identified allelochemicals from *Vulpia myuros*, J. Chem. Ecol. 27, 383–394.
- Anderson R.C., Katz A.J., Anderson M.R. (1978) Allelopathy as a vactor in the success of *Helianthus mollis*. Lam, J. Chem. Ecol. 4, 9.
- Bansal G.L. (1990) Allelopathic potential of Linseed on buttercup (*Ranunculus arvensis* L.), in: Triveldi M.L. Gill B.S., Saini S.S. (Eds.), Plant Science Research in India, New Delhi: Today and Tomorrow Printers and Publishers, pp. 801–805.
- Baziramakenga R., Simard R.R., Leroux G.D. (1994) Effects of benzoic and cinnamic acids on growth, mineral composition, and chlorophyll content of soybean, J. Chem. Ecol. 20, 2821–2833.
- Bell D.B. (1974) The influence of osmotic pressure in tests for allelopathy, T. Illinois State Acad. Sci. 67, 312–317.
- Bell E.A., Charlwood B.V. (1980) Secondary plant products, Encyclopedia of Plant Physiology, New Series, Vol. 8, Springer-Verlag, New York, 674 p.
- Ben-Hammouda M., Kremer R.J., Minor H.C. (1995) Phytotoxicity of extracts from sorghum plant components on wheat seedlings, Crop Sci. 35, 1652–1656.
- Bendall G.M. (1975) The allelopathic activity of California thistle (*Circium arvense*) in Tasmania, Weed Res. 15, 77–81.
- Bewley J.D., Black M. (1994) Seeds: Physiology of development and germination, 2nd ed., Plenum, New York.
- Black M. (1989) Seed research past, present and future, in: Taylorson R.B. (Ed.), Recent advances in the development and germination of seeds. Plenum, New York, pp. 1–6.
- Bode H.R. (1940) Uber die Blattausscheidungen des wermuts and ihre Werkung auf andere Pflanzen, Planta 30, 567–572.
- Carlson J.R. Jr., Ditterline R.L., Martin J.M., Sands D.C., Lund R.E. (1983) Alfalfa seed germination in antibiotic agar containing NaCl, Crop Sci. 23, 882–885.

- Chaves N., Sosa T., Escudero J.C. (2001) Plant growth inhibiting flavonoids in exudate of Cistus ladanifer and in associated soils, J. Chem. Ecol. 27, 623–631.
- Chon S.U. (2004a) Allelopathic potential of common thistle (*Cirsium japonicum*) leaf extracts and residues, Korean J. Weed Sci. 24, 79–86.
- Chon S.U. (2004b) Phytotoxic effect of Xanthium occidentale leaf extracts on seed germination and early seedling growth of alfalfa and barnyard grass, Korean J. Crop Sci. 49, 30–35.
- Chon S.U., Kim. D.K. (2005) Allelopathic potential of *Xanthium occidentale* Extracts and Residues, Korean J. Weed Sci. 25, 163–170.
- Chon S.U., Nelson C.J. (2001) Effects of experimental procedures and conditions on bioassay sensitivity of alfalfa autotoxicity, Commun. Soil Sci. Plan. 32, 1607–1619.
- Chon S.U., Choi S.K., Chung S., Jang H.G., Pyo B.S., Kim S.M. (2002) Effects of alfalfa leaf extracts and phenolic allelochemicals on early seedling growth and root morphology of alfalfa and barnyard grass, Crop Prot. 21, 1077–1082.
- Chon S.U., Coutts J.H., Nelson C.J. (2000) Effects of light, growth media, and seedling orientation on bioassays of alfalfa autotoxicity, Agron. J. 92, 715–720.
- Chon S.U., Coutts J.H., Nelson C.J. (2004) Osmotic and autotoxic effects of leaf extracts on germination and seedling growth of alfalfa, Agron. J. 96, 1673–1679.
- Chon S.U., Jang H.G., Kim D.K., Kim Y.M., Boo H.O., Kim Y.J. (2005) Allelopathic potential in lettuce (*Lactuca sativa* L.) plants, Sci. Hortic. 106, 309–317.
- Chon S.U., Jennings J.A., Nelson C.J. (2004) Alfalfa (*Medicago sativa* L.) autotoxicity: Current status, Allelopathy J. (in print).
- Chon S.U., Kim Y.M., Lee J.C. (2003a) Herbicidal potential and quantification of causative allelochemicals from several compositae weeds, Weed Res. 43, 444–450.
- Chon S.U., Nelson C.J., Coutts J.H. (2003b) Physiological Assessment and Path Coefficient Analysis to Improve Evaluation of Alfalfa Autotoxicity, J. Chem. Ecol. 29, 2395–2406.
- Chou C.H. (1980) Allelopathic researches in the subtropical vegetation in Taiwan, Comp. Physiol. Ecol. 5, 222–234.
- Chou C.H., Leu L.L. (1992) Allelopathic substances and interactions of *Delonix regia* (BOJ) Raf, J. Chem. Ecol. 18, 2285–2303.
- Chung I.M., Miller D.A. (1995a) Differences in autotoxicity among seven alfalfa cultivars, Agron. J. 87, 596–600.
- Chung I.M., Miller D.A. (1995b) Natural herbicide potential of alfalfa residue on selected weed species, Agron. J. 87, 920–925.
- Cochran V.L., Elliott L.F., Papendick R.I. (1980) Carbon and nitrogen movement from surface-applied wheat straw, Soil Sci. Soc. Am. J. 44, 978–982.
- Connell J.H. (1990) Apparent versus "real" competition in plants, in: Grace J.B., Tilman D. (Eds.), Perspectives on plant competition, Academic Press, San Diego, CA, pp. 9–25.
- Cruz-Ortega R., Anaya A.L., Ramos L. (1988) Effects of allelopathic compounds of corn pollen on respiration and cell division of watermelon, J. Chem. Ecol. 14, 71–86.
- Dayan F.E., Romagni J.G., Duke S.O. (2000) Investigation of the mode of action of natural phytotoxins, J. Chem. Ecol. 26, 2079–2094.

- Dhyani S.K. (1978) Allelopathic potential of *Eupatorium* adenophorum on seed germination of *Lantana camara* var. aculeata, Indian J. Forestry 1, 113–119.
- Dilday R.H., Lin J., Yan W. (1994) Identification of allelopathy in the USDA-ARS rice germplasm collection, Aust. J. Exp. Agr. 34, 907–910.
- Dornbos D.L. Jr., Spencer G.F. (1990) Natural products phytotoxicity, A bioassay suitable for small quantities of slightly water-soluble compounds, J. Chem. Ecol. 16, 339–351.
- Duke S.O., Vaughn K.C., Croom E.M. Jr., Elsohly H.N. (1987) Artemisinin, a constituent of annual wormwood (*Artemisia* annua), is a selective phytotoxin, Weed Sci. 35, 499–505.
- Einhellig F.A. (1986) Mechanisms and modes of action of allelochemicals, in: Putnam A.R., Tang C.S. (Eds.), The science of allelopathy, John Wiley & Sons, New York, pp. 171–187.
- Einhellig F.A., Rasmussen J.A. (1993) Effect of root exudates sorgoleone on photosynthesis, J. Chem. Ecol. 19, 369–375.
- Elliott L.F., Cochran V.L., Papendick R.I. (1981) Wheat residue and nitrogen placement effects on wheat growth in the greenhouse, Soil Sci. 131, 48–52.
- Fadayomi O., Oyebade E.O. (1984) An investigation of allellopathy in Siam weed (*Eupatorium odoratum*), Geobios 11, 145–150.
- Fay P.K., Duke W.B (1977) An assessment of allelopathic potential in Avena gemplasm, Weed Sci. 25, 224–228.
- Fennimore S.A., Umeda K. (2003) Weed control in glyphosatetolerant lettuce (*Lactuca sativa*), Weed Technol. 17, 738–746.
- Fick G.W., Holt D.A., Lugg D.G. (1988) Environmental physiology and crop growth, in: Hanson A.A., Barnes D.K., Hill R.R. (Eds.), Alfalfa and alfalfa improvement, Agron. Mono. 29, ASA, CSSA, and SSSA, Madison, WI, pp. 163–194.
- Fisher N.H., Williamson G.B., Weidenhamer J.D., Richardson D.R. (1994) In search of allelopathy in the Florida scrub: The role of terpenoids, J. Chem. Ecol. 20, 1355–1380.
- Fredman J., Rushkin E., Waller G.R. (1982) Highly potent germination inhibitors in aqueous eluate of fruits of bishops weed (*Ammi majus* L.) and avoidance of autoinhibition, J. Chem. Ecol. 8, 55–65.
- Fuerst E.P., Putnam A.R. (1983) Separating the competitive and allelopathic components of interference: Theoretical principles, J. Chem. Ecol. 9, 937–944.
- Funke G.L. (1943) The influence of Artemisia absinthium on neighbouring plants, BLUMEA 5, 281–293.
- Gentle C.B., Duggin J.A. (1997) Allelopathy as a competitive strategy in persistent thickets of *Lanata camara* L. in three Australian forest communities, Plant Ecol. 132, 85–95.
- Goel U., Sareen T.S. (1986) Allelopathic effectof trees on the undersory vegetation, Acta Bot. Indica 14, 162–166.
- Goplen B.P., Webster G.R. (1969) Selection in *Medicago sativa* for tolerance to alfalfa-sick soil of central Alberta, Agron. J. 61, 589–590.
- Haar M.J., Fennimore S.A. (2003) Evaluation of integrated practices for common purslane (*Portulaca oleracea*) management in lettuce (*Lactuca sativa*), Weed Technol. 17, 229–233.
- Hall M.H., Henderlong P.R. (1989) Alfalfa autotoxic fraction characterization and initial separation, Crop Sci. 29, 425–428.
- Halligan J.P. (1976) Toxicity of Artemisia californica to four associated herb species, Am. Midl. Nat. 95, 406–421.

- Harborne J.B. (1980) Plant phenolics, in: Bell E.A., Charlwood B.V. (Eds.), Secondary Plant Products. Encyclopedia of Plant Physiology, New Series, Vol. 8, Springer-Verlag, New York, pp. 329–402.
- Hegarty T.W., Ross H.A. (1978) Differential sensitivity to moisture stress of seed germination and seedling radicle growth in calabrese (*Brassica oleraces* var. *italica*) and cress (*Lepidium sativum*), Ann. Bot. 42, 1003–1005.
- Hegde R.S., Miller D.A. (1992a) Concentration dependency and stage of crop growth in alfalfa autotoxicity, Agron. J. 84, 940–946.
- Hegde R.S., Miller D.A. (1992b) Scanning electron microscopy for studying root morphology and anatomy in alfalfa autotoxicity, Agron. J. 84, 618–620.
- Helgeson E.A., Konzak R. (1950) Phytotoxic effects of aqueous extracts of field bindweed and of Canada thistle. A preliminary report, North Dakota Agric. Exp. Station Bull. 12, 71–76.
- Hodgson J.M. (1963) Canada thistle, you can't afford to kill them, Montana Agric. Exp. Station Circ. 241, 1–2.
- Hodgson J.M. (1968) The nature, ecology and control of Canada thistle, Tech. Bull. No. 1386, Agri. Res. Serve, USDA, USA.
- Holm L. (1978) Some characteristics of weed problems in two worlds, Proc. Western Soc. Weed Sci. 31, 3–12.
- Horsley S.B. (1977) Allelopathic interference among plants. II. Physiological modes of action, Proceedings Fourth North American Forest Biology Workshop, in: Wilcox H.E., Hamer A.F. (Eds.), Syracuse, New York: State University of New York, pp. 93–136.
- Hsiao T.C. (1973) Plant responses to water stress, Ann. Rev. Plant Physiol. 24, 519–570.
- Huang Z., Liao L., Wang S., Cao G. (2000) Allelopathy of phenolics from decomposing stump-roots in replant Chinese fir woodland, J. Chem. Ecol. 26, 2211–2219.
- Inam B., Hussain F., Farhat B. (1987) Allelopathic effects of Pakistani weeds: *Xanthium strumarium* L, Pakistan J. Sci. Ind. Res. 30, 530–533.
- Inam B., Hussain F., Farhat B. (1989) Cannabis sativa L. is allelopathic, Pakistan J. Sci. Ind. Res. 32, 617–620.
- Inderjit (1996) Plant phenolics in allelopathy, Bot. Rev. 62, 186–202.
- Inderjit, Dakshini K.M.M. (1995) On laboratory bioassays in allelopathy, Bot. Rev. 61, 28–44.
- Inderjit, Dakshini K.M.M. (1999) Allelopathy: One component in a multifaceted approach to ecology, in: Inderjit, Dakshini K., Foy C.L. (Eds.), Principles and Practices in Plant Ecology: Allelochemical Interactions, CRC Press, Boca Raton, FL, USA, pp. 3–4.
- Inderjit, Del Moral R. (1997) Is separating resource competition from allelopathy realistic? Bot. Rev. 63, 221–230.
- Inderjit, Olofsdotter M. (1998) Bioassays for rice allelopathy: Some concerns, in: Olofsdotter M. (Ed.), Allelopathy in rice, Int. Rice Res. Inst. Pres Manila, Philippines, pp. 45–55.
- Jackson J.R., Willemsen R.W. (1976) Allelopathy in the first atages of secondary succession on the piedmont of New Jersey, Am. J. Bot. 7, 1015–1020.
- Jennings J.A. (1996) Rotational interval, soil texture and zone of influence studies on alfalfa autotoxicity, Ph.D. Dissertation, University of Missouri, Columbia, MO.
- Jennings J.A., Nelson C.J. (1998) Influence of soil texture on alfalfa autotoxicity, Agron. J. 90, 54–58.

- Kadioglu I. (2000) Effects of hearleaf cocklebur (*Xanthium strumarium* L.) extract on some crops and weeds, Asian J. Plant Sci. 3, 696–700.
- Kanchan S.D., Jayachandra (1979) Allelopathic effects of *Parthenium hysterophorus* L. III. Inhibitory effects of weed residues, Plant Soil 53, 37–47.
- Kaur H., Inderjit, Kaushik S. (2005) Cellular evidence of allelopathic interference of benzoic acid to mustard (*Brassica juncea* L.) seedling growth, Plant Physiol. Biochem. 43, 77–81.
- Kil B.S., Yun K.W. (1992) Allelopathic effects of water extracts of *Artemisia princeps* var. *orientalis* on selected plant species, J. Chem. Ecol. 18, 39–51.
- Kil B.S., Yun K.W., Lee S.Y. (1992) Influence of Artemisia princeps var. orientalis components on callus induction and growth, J. Chem. Ecol. 18, 1455–1462.
- Klein R.R., Miller D.A. (1980) Allelopathy and its role in agriculture, Commun. Soil Sci. Plan. 11, 43–56.
- Kohl R.K., Batish D.R. (1994) Exhibition of allelopathy by *Parthenium hysterophorus* L. in agroecosystems, Trop. Ecol. 35, 295–307.
- Kohli R.K. (1990) Allelopathic Potential of Ecucalyptus, India: MAB-DoEn Project Report.
- Kohli R.K., Daizy R., Verma R.C. (1993) A mathematical model to predict the tissue response to parthenin-an allelochemical, Biol. Plantarum 35, 567–576.
- Kumari A., Kohli R.K. (1987) Autotoxicity of ragweed parthenium Parthenium hysterophorus, Weed Sci. 35, 629–632.
- Leather G.R. (1983) Sunflowers (*Helianthus annus*) are allelopathic to weeds, Weed Sci. 31, 37–41.
- Lee T.T. (1977) Role of phenolic inhibitors in peroxidasemediated degradation of indol-3-acetic acid, Plant Physiol. 59, 372–375.
- Letourneau D., Failes G.D., Heggeness H.G. (1956) The effect of aqueous extracts of plant tissue on germination of seeds and growth of seedlings, Weeds 4, 363–368.
- Liu D.L., Lovett J.V. (1993) Biologically active secondary metabolites of barley. II. Phytotoxicity of barley allelochemicals, J. Chem. Ecol. 19, 2231–2244.
- Luu K.T., Matches A.G., Peters E.J. (1982) Allelopathic effect of tall fescue on birdsfoot trefoil as influenced by N fertilization and seasonal changes, Agron. J. 74, 805–808.
- Mann H.H., Barnes T.W. (1945) The competition between barley and certain weeds under controlled conditions, Ann. Appl. Biol. 32, 15–22.
- Manners G.D., Galitz D.S. (1986) Allelopathy of small everlasting (*Antennaria microphylla*). I. Identification of constituents phytotoxic to leafy spurge (*Euphorbia esula*), Weed Sci. 34, 8–12.
- Markova S.A. (1972) Experimental investigations of the influence of oats on growth and development of *Erysimum cheiranthoides* L., Physiological-Biochemical Basis of Plant Interactions in Phytocenoses, in: Grodzinsky A.M. (Ed.), Vol. 3, pp. 66–68, Naukova Dumka, Kiev (in Russian, English summary).
- Meepagala K.M., Kuhajek J.M., Sturtz G.D., Wedge D.E. (2003) Vulgarone B, the antifungal constituent in the steam-distilled fraction of *Artemisia douglasiana*, J. Chem. Ecol. 29, 1771–1780.
- Mersie W., Singh M. (1978) Allelopathic effects of parthenium (Parthenium hysterophorus L.) extract and residue

on some agronomic crops and weeds, J. Chem. Ecol. 13, 1739–1747.

- Miller D.A. (1983) Allelopathic effects of alfalfa, J. Chem. Ecol. 9, 1059–1072.
- Miller D.A. (1992) Allelopathy and establishment, Alfalfa Talk, Vol. 12, No. 1, Certified Alfalfa Seed Council, Davis, CA.
- Miller D.A. (1996) Allelopathy in forage crop systems, Agron. J. 88, 854–859.
- Miller R.W., Kleiman R., Powell R.G., Putnam A.R. (1988) Germination and growth inhibitors of alfalfa, J. Nat. Prod. 51, 328–330.
- Molisch H. (1937) Der Einfluss einer Pflanze auf die andere -Allelopathie, Fischer, Jena, Germany.
- Neill R.L., Rice E.L. (1971) Possible role of *Ambrosia psilostachya* on patterning and succession in old fields, Am. Midl. Nat. 86, 344–350.
- Newby V.K., Sablon R.M., Synge R.L. (1980) Free and bound phenolic acids of lucerne (*Medicago sativa* cv. Europe), Phytochemistry 19, 651–657.
- Olofsdotter M., Navarez D. (1996) Allelopathic rice for Echinoch-loa crus-galli control, Proc. Int. Wide Control Conf., 2nd ed., Slagelse, Denmark, 25–28 June 1996, Dep. of Weed Control and Pestic. Ecology, Slagelse, Denmark, pp. 1175–1181.
- Park B.K. (1966) A study on productivity of grassland in Korea (I), Theses Collection of Institute of Korean Culture, Ewha Womens University 8, pp. 81–90.
- Pederson G.A. (1986) White clover seed germination in agar containing tall fescue leaf extracts, Crop Sci. 26, 1248–1249.
- Peters E.J. (1968) Toxicity of tall fescue to rape and birdsfoot trefoil seeds and seedlings, Crop Sci. 8, 650–653.
- Putnam A.R., Duke W.B. (1974) Biological suppression of weeds: Evidence for allelopathy in accessions cucumber, Science 185, 370–372.
- Rasmussen J.A., Einhellig F.A. (1979) Allelopathic effects of leaf extracts of *Ambrosia trifida* (Compositae), Southwest. Nat. 24, 637–644.
- Rice E.L. (1974) Allelopathy, Academic Press, New York, 1.
- Rice E.L. (1984) Allelopathy, 2nd ed., Academic Press, New York, 422 p.
- Rice E.L. (1986) The Science of Allelopathy, in: Putnam A.R., Tang C.S. (Eds.), Wiley: New York, pp. 23–42.
- Rice E.L. (1995) Biological control of weeds and plant diseases: Advances in applied allelopathy, Univ. of Oklahoma Press, Norman.
- Ridenour W.M., Callaway R.M. (2001) The relative importance of allelopathy in interference: the effects of an invasive weed on a native bunchgrass, Oecologia 126, 444–450.
- Santos B.M., Dusky J.A., Stall W.M., Bewick T.A., Shilling D.G. (2003) Mechanisms of interference of smooth pigweed (*Amaranthus hybridus*) and common purslane (*Portulaca oleracea*) on lettuce as influenced by phosphorus fertility, Weed Sci. 52, 78–82.
- Sarma K.K.V. (1976) Allelopathic potential of *Parthenium hysterophorus* L. on seed germination, and dry matter production in *Arachis hypogea* Willd., *Crotoloria juncea* L., and *Phaseolus mungo* Linn, Trop. Ecol. 17, 76–82.
- Siegler D.S. (1996) Chemistry and mechanisms of allelopathic interactions, Agron. J. 88, 876–885.
- Slatyer R.O. (1967) Plant-water relations, Academic Press, New York.

- Stachon W.J., Zimdal R.L. (1980) Allelopathic activitity of Canada Thistle (*Circium arvense*) in Colorado, Weed Sci. 28, 83–86.
- Stairs A.F. (1986) Life history variation in Artemisia campestris on Lake Huron sand dune system. Ph.D. thesis, The University of Western Ontario, London, Ontario.
- Stevens G.A., Tang C.S. (1985) Inhibition of seedling growth of crop species by recirculating root exudates of *Bidens pilosa*, J. Chem. Ecol. 11, 166–167.
- Stevens K.L. (1986) Allelopathic polyacetlyenes from *Centaurea repens* (Russian knapwed), J. Chem. Ecol. 12, 1205–1211.
- Stevens K.L., Merrill G.B. (1985) Sesquiterpene lactones and allelochemicals from Centaurea species, in: Thompson A.C. (Ed.), The Chemistry of Allelopathy: Biochemical Interactions Among Plants, American Chemical Society, Washington, DC, USA, pp. 83–98.
- Tan R.X., Zheng W.F., Tang H.Q. (1998) Biologically active substances from the genus Artemisia, Planta Med. 64, 295–302.
- Thijs H., Shann J.R., Weidenhamer J.D. (1994) The effect of phytotoxins on competitive outcome in a model system, Ecology 75, 1959–1964.
- Trenbath B.R. (1974) Biomass productivity of mixtures, Adv. Agron. 26, 177–210.
- Van Sumere C.F., Cottenie J., Degreef J., Kint J. (1972) Biochemical studies in relation to the possible germination regulatory role of naturally occurring coumarin and phenolics, Recent Adv. Phytochem. 4, 165–221.
- Wardle D.A., Nicholson K.S., Ahmed M. (1992) Comparison of osmotic and allelopathic effects of grass leaf extracts on grass seed germination and radicle elongation, Plant Soil 140, 315–319.

- Weidenhamer J.D. (1996) Distinguishing resource competition and chemical interference: Overcoming the methodological impasse, Agron J. 88, 866–875.
- Weidenhamer J.D., Morton T.C., Romeo J.T. (1987) Solution volume and seed number: Often overlooked factors in allelopathic bioassay, J. Chem. Ecol. 13, 1481–1491.
- Wilson R.G. Jr. (1981) Effect of Canada thistle (*Cirsium arvense*) residues on growth of some crops, Weed Sci. 29, 159–165.
- Wu H., Haig T., Pratley J., Lemerle D., An M. (2001) Allelochemicals in wheat (*Triticum aestivum* L.): Variation of phemolic acids in shoot tissues, J. Chem. Ecol. 27, 125–135.
- Wu H., Pratley J., Lemerle D., Haig T. (1999) Identification and evaluation of toxicity of rice root elongation inhibitors in flooded soils with added wheat straw, Soil Sci. Plant Nutr. 36, 97–103.
- Yun K.W. (1991) Allelopathic effects of chemical substances in Artemisia princeps var. orientalis on selected species, Ph.D. thesis, Wonkwang University, Iri, Korea.
- Yun K.W., Choi S.K. (2003) Seasonal variation in allelopathic potential of *Artemisia princeps* var. *orientalis* on plants and microbes, J. Plant Biol. 46, 105–110.
- Yun K.W., Kil B.S. (1992) Assessment of allelopathic potential in *Artemisia princes* var. *orientalis* residues, J. Chem. Ecol. 18, 1933–1940.
- Yun K.W., Maun M.A. (1997) Allelopathic potential of Artemisia campestris ssp. caudata on Lake Huron sand dunes, Can. J. Bot. 75, 1903–1912.
- Yun K.W., Kil B.S., Han D.M. (1993) Phytotoxic and antimicrobial activity of volatile constituents of *Artemisia princeps* var. *orientalis*, J. Chem. Ecol. 19, 2757–2766.

# Part VI Soil Health

## **Assessing the Productivity Function of Soils**

Lothar Mueller, Uwe Schindler, Wilfried Mirschel, T. Graham Shepherd, Bruce C. Ball, Katharina Helming, Jutta Rogasik, Frank Eulenstein, and Hubert Wiggering

#### Contents

1	Intro on th	oduction - The Demand for Information ne Productivity Function of Soils	744
2	Soils	and Their Constraints to Plant Growth	745
3	Infor Syste	mation on Taxonomic Soil Classification ems for Soil Productivity Potentials	746
4	Soil ( Qual	Structure as a Criterion of Agricultural Soil ity	746
5	Meth Func 5.1	nods of Assessing the Overall Productivity           ction of Soil           Soil and Land Evaluation in a Historical	747
	5.2	Context	747 748
	5.3	Comparison of Methods of Soil Evaluation Relevant to Soil Productivity	752
6	Targ the S	ets and Steps to Assessing foil Productivity Function	
	in th	e 21st Century	753
7	Conc	clusions	754
Re	eferen	ces	755

e-mail: mueller@zalf.de

Abstract The development and survival or disappearance of civilizations has been based on the performance of soils to provide food, fibre, and further essential goods for humans. Amongst soil functions, the capacity to produce plant biomass (productivity function) remains essential. This function is closely associated with the main global issues of the 21st century like food security, demands of energy and water, carbon balance and climate change. A standardised methodology for assessing the productivity function of the global soil resource consistently over different spatial scales will be demanded by a growing international community of land users and stakeholders for achieving high soil productivity in the context of sustainable multifunctional use of soils. We analysed available methods for assessing the soil productivity function. The aim was to find potentials, deficiencies and gaps in knowledge of current approaches towards a global reference framework. Our main findings were (i) that the soil moisture and thermal regime, which are climate-influenced, are the main constraints to the soil productivity potential on a global scale, and (ii) that most taxonomic soil classification systems including the World Reference Basis for Soil Resources provide little information on soil functionality in particular the productivity function. We found (iii) a multitude of approaches developed at the national and local scale in the last century for assessing mainly specific aspects of potential soil and land productivity. Their soil data inputs differ, evaluation ratings are not transferable and thus not applicable in international and global studies. At an international level or global scale, methods like agro-ecological zoning or ecosystem and crop modelling provide assessments of land

L. Mueller (🖂)

Leibniz-Zentrum für Agrarlandschaftsforschung (ZALF) Müncheberg, Eberswalder Straße 84, 15374 Müncheberg, Germany

productivity but contain little soil information. Those methods are not intended for field scale application to detect main soil constraints and thereby to derive soil management and conservation recommendations in situ. We found also that (iv) soil structure is a crucial criterion of agricultural soil quality and methods of visual soil assessment like the Peerlkamp scheme, the French method "Le profil cultural" and the New Zealand Visual Soil Assessment are powerful tools for recognising dynamic agricultural soil quality and controlling soil management processes at field scale. We concluded that these approaches have potential to be integrated into an internationally applicable assessment framework of the soil's productivity function, working from field scale to the global level. This framework needs to serve as a reference base for ranking soil productivity potentials on a global scale and as an operational tool for controlling further soil degradation and desertification. Methods like the multi-indicator-based Muencheberg Soil Quality Rating meet most criteria of such a framework. This method has potential to act as a global overall assessment method of the soil productivity function for cropping land and pastoral grassland but needs further evolution by testing and amending its indicator thresholds.

**Keywords** Soil functions • Soil productivity • Soil quality • Soil structure • Soil classification • Sustainable agriculture • Land rating

## 1 Introduction - The Demand for Information on the Productivity Function of Soils

Soils cover most lands of the earth, but regarding their service for humans they are a limited and largely non-renewable resource (Blum, 2006). On the globe about 3.2 billion hectares are used as arable land, which is about a quarter of the total land area (Scherr, 1999; Davis and Masten, 2003). Total agricultural land covers about 40–50% of the global land area (Smith et al., 2007).

The development and survival of civilizations has been based on the performance of soils on this land to provide food and further essential goods for humans (Hillel, 2009). Global issues of the 21st century like food security, demands of energy and water, climate change and biodiversity are associated with the sustainable use of soils (Lal, 2008, 2009; Jones et al., 2009; Lichtfouse et al., 2009). Feeding about 10 billion people is one of the greatest challenges of our century. Borlaug (2007) stated: "The battle to alleviate poverty and improve human health and productivity will require dynamic agricultural development". There are serious concerns that increases of global cereal yield trends are not fast enough to meet expected demands (Cassman et al., 2003). However, agricultural development cannot be intensified regardless of the bearing capacity of soils, ecosystems and socio-economical environment. It has to be imbedded within balanced strategies to develop multi-functional landscapes on our planet (Wiggering et al., 2006; Helming et al., 2008). Handling of soils by societies must be in a sustainable way in order to maintain the function of all global ecosystems (Rao and Rogers, 2006; Ceotto, 2008, Hillel, 2009). This includes the use of soils by agriculture for high productivity (Lal, 2009; Walter and Stützel, 2009). Global carbon, water and nutrient cycles are also affected by agriculture (Bondeau et al., 2007).

Soils have to provide several ecological and social functions (Blum, 1993; Tóth G et al., 2007; Lal, 2008; Jones et al., 2009). Based on a definition of Blum (1993), one of the six key soil functions is "food and other biomass production". The soil protection strategy of the European Commission (EC, 2006; Tóth G et al., 2007) addresses "biomass production" as a main soil function which must be maintained sustainably. We call this the "productivity function". The productivity function is related to the most common definition of soil quality as "the capacity of a specific kind of soil to function, within natural or managed ecosystem boundaries, to sustain plant and animal productivity, maintain or enhance water and air quality, and support human health and habitation" (Karlen et al., 1997). Based on this definition, the objective comes close to the assessment of "agricultural soil quality".

Although the productivity function of soils is of crucial importance, it is sometimes ill-defined or its description may be very different. In the German soil protection Act (BBodSchG, 1998) the productivity function is about "utility for agriculture and forestry". Amongst those utility functions (agriculture, resources, settlement and traffic), soils used by agriculture and forestry have a unique position.

Firstly, agricultural soils have to be used sustainably to maintain their productivity potential long-term. Secondly, natural soil functions (habitat, nutrient cycling, biofiltering) are not only the domain of soils in natural protected areas. Agricultural soils have to fulfil their natural functions too, e.g. provide or support ecosystem services (Foley et al., 2005). Assessing the productivity function is not restricted to specific land use concepts with regard to management intensity. It embraces the capacity of soils for low- input and organic farming approaches. Also, soils in more natural ecosystems may provide some productivity function. This chapter focuses on the productivity function of soil on agricultural land. We shall analyse available methods and tools for assessing the state of soils concerning their ability to provide the productivity function. We consider which evaluation tools are available to quantify soil productivity and which tools are needed to meet further demands under changing climate and soil management. We start from the hypothesis that a growing community of land users and stakeholders has to achieve a high productivity without any significant detrimental long-term impact on soils and the environment. This requires an increasing awareness of a demand to assess the productivity of their soils using internationally standardised frameworks and simple diagnostic tools.

Our focus shall be on answering the following questions:

- Which properties of soils most affect their productivity?
- Which information on soil productivity potentials do existing soil classification systems provide?
- What methods of assessing the productivity function of soils are available?
- How useful are these methods in assessing different aspects of agricultural soil quality?

Conclusions are made for the development of a framework and evaluation tools of agricultural soil quality consistently over different scales as a basis for monitoring and sustainable management of soils.

## 2 Soils and Their Constraints to Plant Growth

Soils are components of terrestrial ecosystems. The productivity of these systems is controlled by natural

factors and by human activity. Most important external natural factors are solar radiation, influencing temperature and evapotranspiration, and/or precipitation (Lieth, 1975). Soils may provide for plant growth if climate, as the main soil forming factor, is in an appropriate range (Murray et al., 1983, Lavalle et al., 2009). Thus, on a global scale, natural constraints to soil productivity can be classified into three major groups. The first group includes the thermal and moisture regimes of soils. Plants require appropriate soil temperatures and moisture for their growth (Murray et al., 1983; Lavalle et al., 2009). For most soils, thermal and moisture regimes are directly dependent on climatic conditions. They define the frame for limitations like drought, wetness, or a too short vegetation period, limiting the productivity (Fischer et al., 2002).

Worldwide, soil moisture is the main limiting factor in most agricultural systems (Hillel and Rosenzweig, 2002; Debaeke and Aboudrare, 2004; Ciais et al., 2005; Verhulst et al., 2009; Farooq et al., 2009). Drylands cover more than 50% of the global land surface (Asner and Heidebrecht, 2005). Available soil water is a prerequisite for plant growth. In all climates suitable for agriculture, the water storage capacity of soils is a crucial property for soil functionality including the productivity function (AG Boden, 2005; Shaxson, 2006; Jones et al., 2009). It is closely correlated with crop yields (Harrach, 1982; Wong and Asseng, 2006).

The second group of restrictions includes other internal soil deficiencies mainly due to an improper substratum limiting rooting and nutrition of plants. These include shallow soils, stoniness, hard pans, anaerobic horizons, or soils with adverse chemistry such as salinity, sodicity, acidity, nutrient depletion or contamination which may cause severe restrictions to plant growth or the utilisation of biomass (Murray et al., 1983; Louwagie et al., 2009).

The third group includes topography, sometimes considered as an external soil property, preventing soil erosion and providing accessibility by humans and machinery (Fischer et al., 2002; Duran Zuazo, 2008).

There seems to be an interaction between natural constraints to soil productivity and societal factors. Historically, many countries with poor soils tended to be poorly developed. This has led to accelerated soil degradation. Currently, in developing countries, about two thirds of soils have severe constraints to agriculture. Their low fertility (38%), sandy or stony

soils (23%), poor soil drainage (20%) and steep slopes (10%) are the main limits to productivity (Scherr, 1999).

## 3 Information on Taxonomic Soil Classification Systems for Soil Productivity Potentials

Soil classification systems are based on a combination of different criteria. Attributes used for classification may reflect both pedogenesis and pedofunction (Schroeder and Lamp, 1976; Beinroth and Stahr, 2005). Whilst morphological and functional criteria dominated soil classification until the 19th century, pedogenic criteria prevail at higher levels in national soil classification systems since the 20th century (Ahrens et al., 2002; Beinroth and Stahr, 2005). Functional information like the type of substrate is also part of most current soil classifications. In some cases pedogenic and functional criteria are combined, and genetic soil types provide information about soil productivity potentials. For example, Chernozems, which have developed mainly from loessial material and have a mollic epipedon, rich in humus, have a high crop yield potential, whilst Leptosols are shallow soils of low productivity. Podzols are leached sandy soils lacking nutrients and water storage capacity. These examples show that if the soil type or reference soil group is associated with typical substrate and climate conditions, some functional properties may be determinable.

Apart from these extremes, functional information derivable from higher level soil classifications is relatively low. Some soil types or reference soil groups such as Cambisols, Fluvisols or Regosols may have developed from different soil substrates in different climatic environments. In those cases, more relevant information about possible soil productivity at a local or regional scale is provided if the classification includes further soil attributes like texture, organic matter, degree of trophy and pH. Soil texture is correlated with other important functional attributes like water and nutrient storage capacity and thus has become a dominant criterion of all existing functional classification systems since soil began to be managed (Storie, 1933; Rothkegel, 1950; Feller et al., 2003, Feller et al., 2003; Beinroth and Stahr, 2005; Begon et al., 2006).

As the USDA soil classification (Keys to Soil Taxonomy, 2006) includes climate information in terms of soil moisture and temperature regime classes, correlations of soils with their productivity at a hierarchy level of great groups (3rd level) are relatively high. In contrast, the FAO soil map of the world and the latest reference base for soil resources (WRB, 2006) lack information about temperature and moisture regimes and thus information on soil productivity potentials. For a rough assessment of soil productivity potentials in Africa, Eswaran et al. (1997), had to translate the FAO soil map of Africa into the USDA soil taxonomy by supplementing climate information.

At the lowest levels of the soil classification hierarchy, functional information on particular soils is greatest. Soils classified at series level in USDA Soil Taxonomy, in the UK soil classification, or local soil types on forest sites in some federal states of Germany, contain detailed information on soil morphological and functional properties, which can be linked with soil productivity data (Mausel et al., 1975; Kopp and Schwanecke, 2003). However, the specific data and correlations cannot be transferred to other regions.

Soil taxonomic classifications sometimes include information on soil structure, which often reflects anthropogenic impacts within human timescales on soil. This information provision can be relatively high with some soils like Histosols in the AG Boden (2005), and Keys to Soil Taxonomy (2006), but it is low with most mineral soils.

## 4 Soil Structure as a Criterion of Agricultural Soil Quality

Soil structure is a complex category and a key to soil biological, chemical and physical processes (Jackson et al., 2003; Karlen, 2004; Bronick and Lal 2005; Kay et al., 2006; Roger-Estrade et al., 2009). The spatial arrangement of aggregates and porosity is a main aspect of soil structure. Structure is related to soil function, e.g. to the productivity function or to water and solute transport. Unfavourable structure can result in lower crop yields and greater leaching losses (Kavdir and Smucker, 2005). Current structure features and function result from soil substratum, genetic

and management factors. Soil structure is vulnerable to change by compaction and erosion and its preservation is key to sustaining soil function. Crop rotation and tillage strategies should aim to produce optimum soil structure for high and sustainable crop yields (Hulugalle et al., 2007). A good soil structure for plant growth may play a particularly important role in organic farming while poor soil structure cannot be compensated by an extra input of agrochemicals in those systems (Munkholm et al., 2003).

Visible soil structure revealed by digging up the soil shows the abundance and arrangement of soil aggregates and roots which may indicate properties of soils that are dependent on soil management (Shepherd, 2000; McKenzie, 2001; Lin et al., 2005; Mueller et al., 2009). It reflects important aspects of the dynamic indicators of soil quality, indicators that can be categorised and used to monitor and control the status of soil. Farmers and gardeners do this in an individual, experienced-based visual-tactile manner. Visual-tactile recognizable soil features like colour, texture, moisture conditions, earthworm casts may serve to evaluate and classify the quality of soil (Shaxson, 2006).

As indigenous people have done before, soil science and soil advisory services utilise the same common field diagnostic criteria within defined frameworks and check their validity over larger scales. Over the past decades, the interest in soil structure evaluation as a diagnostic tool for assessments of dynamic, e.g management-induced, soil quality has been recognised and has evolved (Shepherd, 2000; McKenzie, 2001; Lin et al., 2005; Shaxson, 2006). Methods of visual soil structure examination enable semi- quantitative information for use in extension and monitoring (Shepherd, 2000; McKenzie, 2001) or even modeling (Roger-Estrade et al., 2004, 2009). One of their advantages is a quick, reliable assessment of good, acceptable or poor states of soil structure. Soil structural features meet the farmer's perception on soil quality (Shepherd, 2000; Batey and Mc Kenzie 2006) and are correlated with measured data of physical soil quality (Lin et al., 2005) and crop yield (Mueller et al., 2009). However, clearly defined rules and scoring methods are necessary to minimise subjective errors.

Several methods have been developed over the past five decades. One of the oldest but most accepted methods is that of Peerlkamp (1967). The traditional French method "Le profil cultural" (Roger-Estrade et al., 2004) belongs to a group of more sophisticated methods providing detailed information on the total soil profile. A quantitative comparison of some methods and their correlations with measured physical parameters after standardizing data revealed that most methods provided similar results (Mueller et al., 2009). Types and sizes of aggregates and abundance of biological macropores were the most reliable criteria as related to measurement data and crop yields. Differences in soil management could be recognised by visual structure criteria (Mueller et al., 2009). Unfavourable visual structure was associated with increased dry bulk density, higher soil strength and lower infiltration rate but correlations were site-specific. Effects of compaction may be detected by visual examination of the soil (Batey and Mc Kenzie 2006).

Visual methods based on, or supplemented by illustrations, have clear advantages for the reliable assignment of a rating score based on visual diagnostic criteria. The latest development of the Peerlkamp method provided by Ball et al. (2007) is well illustrated (Fig. 1). Also, the New Zealand Visual Soil Assessment (VSA, Shepherd, 2000, 2009) as an illustrated multi-criteria method, enables reliable assessments of the soil structure status. These are feasible tools for structure monitoring and management recommendations. However, they may explain only part of crop yield variability, as the influence of inherent soil properties and climate on crop yield is dominant, particularly over larger regions.

### 5 Methods of Assessing the Overall Productivity Function of Soil

## 5.1 Soil and Land Evaluation in a Historical Context

In a global context, the utilisation of the soil productivity function in agriculture requires not only soils but also an appropriate climate and human activity. Methods for the evaluation of the potential for the productivity of soil have recently been called "land" evaluation methods. "Land evaluation" has been defined as "the process of assessment of land performance when used for specific purposes (FAO, 1976). Historically, land evaluation has developed from soil science. As soil is the most important component of the land resource,

Structure quality	Ease of break up (moist soil)	Size and appearance of aggregates	Visible porosity	Roots	Appearance after break-up: various soils	Appearance after break-up: same soil different tillage	Distinguishing feature	
Sq1 Friable (tends to fall off the spade)	Aggregates readily crumble with fingers	Mostly < 6 mm after crumbling	Highly porous	Roots throughout the soil			Fine aggregates	
Sq2 Intact (most is retained on the spade)	Aggregates easy to break with one hand	A mixture of porous, rounded aggregates from 2mm - 7 cm. No clods present	Most aggregates are porous	Roots throughout the soil			High aggregate porosity	
Sq3 Firm	Most aggregates break with one hand	A mixture of porous aggregates from 2mm -10 cm; less than 30% are <1 cm. Some angular, non-porous aggregates (clods) may be present	Macropores and cracks present. Some porosity within aggregates shown as pores or roots.	Most roots are around aggregates			Low aggregate porosity	1
Sq4 Compact	Requires considerable effort to break aggregates with one hand	Mostly large > 10 cm and sub- angular non- porous; horizontal/platy also possible; less than 30% are <7 cm	Few macropores and cracks	All roots are clustered in macropores and around aggregates			Distinct macropores	1
Sq5 Very compact	Difficult	Mostly large > 10 cm, very few < 7 cm, angular and non-porous	Very low; macropores may be present; may contain anaerobic zones	Few, if any, restricted to cracks			Grey-blue colour	

**Fig. 1** Revised Peerlkamp scale as an example of soil structure evaluation (Ball et al., 2007). The evaluation focuses on aggregates, porosity and roots. Photographs enable a reliable alloca-

tion of scores to real visible features of the topsoil. Intermediate scores and layers of differing scores are possible

soil evaluation is crucial for land evaluation (Rossiter, 1996). In many cases, there is no clear differentiation between soil and land evaluation (Van Diepen et al. 1991; Van de Steeg, 2003). Climate as a main precondition for the production of plant biomass varies over larger spatial scales than soil. Approaches to evaluate the productivity potential of soils from a more regional perspective in similar climates (fields, agricultural regions, smaller countries) tend to prefer the term "soil" for their object of assessment and rating. Approaches coming from a more global perspective (globe, continents, larger countries) tend to emphasise the role of climate and humans in biomass production and favour the term "land". The latter became dominant over the past 40 years, whilst evaluations of the productivity potential of "soil" have a long history, beginning with farming and animal husbandry. Ahrens et al. (2002) stated "... pedology and soil science in general have their rudimentary beginnings in attempts to group or classify soils on the basis of productivity. Early agrarian civilizations must have had some way to communicate differences and similarities among soils." At the beginning of the 19th century the German agronomist A. D. Thaer created a 100 point rating system for the productivity potential of soils based on texture, lime and humus content (Feller et al., 2003). It is one example of a predecessor for some of our current evaluation schemes of agricultural soil quality (Gavrilyuk, 1974; Feller et al., 2003).

#### 5.2 Methods of Soil and Land Rating

#### 5.2.1 Traditional National Soil Ratings

At national level, specific methods for the evaluation and classification of the productivity potential of soils and land have been developed. In Europe they have existed for about 60-100 years. In many countries they are defined by acts of government, have been done by soil surveys and have a high coverage in terms of mapped areas. Examples of those well known soil and land productivity rating systems at national levels are the Storie Index Rating (Storie, 1933), the German and Austrian Soil Rating (German term "Bodenschaetzung"), (Rothkegel, 1950; Pehamberger, 1992; AG Boden, 2005) and the system of soil rating of the former Soviet Union (Gavrilyuk, 1974). These methods try to cover the overall agricultural land with 100% coverage in some countries and are still applied for different purposes, ranging from land taxation to soil protection planning (Hartmann et al., 1999; Preetz, 2003; Rust, 2006). Ratings of these systems have a 100 point scheme in many cases. Data are ordinally scaled. Some methods have been updated and adapted to altered conditions. A main reason was to provide better correlations with current crop yields. The Austrian Soil Rating was amended by climate factors (Bodenaufnahmesysteme in Österreich 2001), whilst other systems like the German Soil Rating have remained unchanged for about 80 years.

#### 5.2.2 More Recent Land Evaluation Systems at National Levels

Over the past 20 years, specific soil and land evaluation systems have been developed or are under construction. Examples of these systems are the US LESA system (Pease and Coughlin, 1996) and the Canadian Land Suitability Rating System for Agricultural Crops (LSRS, Agronomic Interpretations Working Group, 1995). The LESA system consists of a soil evaluation component (Storie Rating) and other factors that contribute to the suitability of land for agriculture, like location, surrounding use and infrastructure. The LSRS system is mainly based on soil attributes and climate factors (Agronomic Interpretations Working Group, 1995). Other countries with substantial agricultural production and fast growing demands like China and Brazil intend to implement quantitative evaluation systems of soil and land productivity (Peng et al., 2002; Bacic et al., 2003; Van de Steeg, 2003; Zhang et al., 2004). Also in Russia there are efforts to establish contemporary soil and land information and evaluation systems (Karmanov et al., 2002; Yakovlev et al., 2006). In the Ukraine, Medvedev et al. (2002) developed an evaluation system of the suitability of land for growing cereals based on soil information and climate data. In Hungary, a modern land evaluation system is being established, containing on-line soil evaluation, which is based on the real-time calculation of D-e-Meter soil fertility index using GIS to produce soil maps at a scale of 1:10 000 (Tóth T. et al., 2007).

All these soil and land evaluation systems are specific in approach, data and scale and their outputs are not or only rarely comparable. Approaches that have been developed for larger countries cover a broader variability of soils and climate and seem to have a better potential for evaluation of agricultural soil quality in trans-national studies.

#### 5.2.3 Soil Capability and Suitability Classifications

Besides productivity ratings, in many countries, classifications of agricultural land limitations (steep lands, dry lands, stony lands), or final allocations to categories like "prime farmland" have been mapped. Examples of those national soil and land capability classifications are the US capability classification (Klingebiel and Montgomery, 1961; Helms, 1992), the UK system developed by the Macaulay Land Use Research Institute (Bibby et al., 1991), the New Zealand land use capability system (Lynn et al., 2009) and the soil fertility classes for agriculture in Australia (Hall, 2008).

Those capability classes are nominal, categorical data, useful for land use planning but not for more detailed productivity assessments within these categories. Data of modern national or federal state soil and land information systems provide tailored medium scale capability classifications.

Soil suitability classifications express soil productivity potentials in terms of the possibility of growing specific crops. In the nineteenth century in German states, soil suitability classification systems using classes ranging from "Prime wheat soil" to "Rye soil" or "Oats soil" were common, and were based on work of Thaer and others (Meyers Lexikon, 1925). As requirements of plants regarding the functional status of soil may differ, all recent soil productivity relevant classifications must have a certain stratification or orientation on crops or groups of crops. Cereals are a basic source of human food supply and while they reflect differences in agricultural soil quality, some systems (Rothkegel, 1950; Agronomic Interpretations Working Group, 1995; Mueller et al., 2007) refer to cereals or cereal-dominated rotations. In the UK, soil suitability classifications have been developed for specific purposes such as direct drilling or reduced tillage. Such systems emphasise the limitations of soil structure and drainage status (Cannell et al., 1978). The presence of climatic data within land use capability classification systems means that such systems can accommodate climate parameters projected into the future. Thus climate change scenarios can be used to identify future changes in land capability (Brown et al., 2008).

#### 5.2.4 Global and Large Regional Soil and Land Evaluations and Classifications

The concept of agro-ecological zoning (AEZ) was developed by the International Institute for Applied Systems Analysis (IIASA) and the FAO (Fischer and Sun, 2001). This sophisticated methodology and model provide a framework for the characterization of climate, soil, and terrain conditions relevant to agricultural production. GIS-based suitability classes for estimating specific crops and their yields over the globe have been calculated and mapped from the sub-national to the global level (Fischer et al., 2002). The system processes soil information, including the FAO/UNESCO Digital Soil Map of the World, with climate information playing the most important role.

The Fertility Capability Classification (FCC Buol et al., 1975) is based on soil survey data and aims to make soil management recommendations and crop yield interpretations. It focuses on those properties and data of soils, topsoils in particular, that are important to fertility management (Sanchez et al., 1982). The system has been mainly applied to the tropics (Sanchez et al., 2003) and updated to a global soil functional capacity classification, providing overviews on single soil constraints to productivity like waterlogging, erosion risk, salinity and others. The basis of both the AEZ methodology and the FCC system are low resolution maps and a limited set of soil parameters and data.

Computer aided land evaluation and classification systems provide capability assessments. MicroLEIS (De la Rosa, 2005) is a system of agro-ecological land evaluation and interpretation of land resources and agricultural management. It has been extended to a decision support system, providing a multifunctional evaluation of soil quality using soil survey input data (De la Rosa et al., 2009).

Crop productivity estimators (Tang et al., 1992) can also be used as research tools and in planning studies. They combine both quantitative and qualitative data to estimate attainable crop yield for different soil units (Verdoodt and van Ranst, 2006). Examples of productivity models with focus on soil erosion are the Productivity Index (PI) model (Pierce et al., 1983), its modifications (Mulengera and Payton, 1999; Duan et al., 2009) and the Erosion Productivity Impact Calculator, EPIC (Williams et al., 1983; Flach, 1986).

The Muencheberg Soil Quality Rating (M-SQR, Mueller et al., 2007) has been developed as a potential international reference base for a functional assessment and classification of soils (Fig. 2). It focuses on cropland and grassland and is based on productivityrelevant indicator ratings which provide a functional coding of soils. Two types of indicator are identified. The first are basic and relate mainly to soil textural and structural properties relevant to plant growth. The second are hazard, relating to severe restrictions of soil function. The sum of weighted basic indicator ratings and multipliers derived from ratings of the most severe (active) hazard indicator yield an overall soil quality rating index. Indicator ratings are based on a field manual and utilize soil survey classifications (AG Boden, 2005; FAO, 2006), soil structure diagnosis tools, and local or regional climate data.

#### 5.2.5 Models Predicting Biomass

There are a large and fast growing number of crop growth and ecosystem models that estimate the local productivity for specific crops, soils and weather data. Models are specific in purpose, vary in their spatial and local scale of resolution, in their focus on particular plants or land use systems, in their proportion and attributes of soil information data and other criteria. These crop growth models can be utilised for assessing the soil productivity for regions where yield data bases exist and the models were parameterised and validated.

On a global scale, modelling climate change relevant issues like possible shortfalls in food production (Tan and Shibasaki, 2003), drought risk



**Fig. 2** Indicator system of the Muencheberg Soil Quality Rating (Mueller et al., 2007). Indicator ratings of soil states are based on rating tables given in a field manual which also contains, where

relevant, hazard indicators and their thresholds. Best soils for cropping and grazing do not have values of hazard indicators which exceed the thresholds

(Alcamo et al., 2007), carbon balance (Bondeau et al., 2007) or GHG emissions (Stehfest et al., 2007) requires reliable calculations of the terrestrial biomass, crop growth and yield. Terrestrial biogeochemical models like the Global Assessment of Security (GLASS) model (Alcamo et al., 2007) containing the Global Agro-Ecological Zones methodology of Fischer et al. (2002) may provide this. Models of this group are valid on a global scale, but the spatial resolution is relatively low. They are sophisticated research tools, not designed for local scale calculations or even management decisions in agriculture.

On a daily temporal basis and local scale working crop production and ecosystem models like DAISY (Hansen et al., 1990), the CERES model family (Ritchie and Godwin, 1993; Xiong et al., 2008), WOFOST (Supit et al., 1994; Hijmans et al., 1994; Reidsma et al., 2009), CANDY (Franko et al., 1995), AGROTOOL (Poluektov et al., 2002), SIMWASER (Stenitzer and Murer, 2003), THESEUS (Wegehenkel et al., 2004), the AGROSIM model family (Mirschel and Wenkel, 2007), DAYCENT (Del Grosso et al., 2005), HERMES (Kersebaum, 2007, 2008) and many others provide productivity estimates of sites under varying conditions of weather, soil moisture or even soil management status.

Models of this group have in common that they are sophisticated and specific from methodology and design to their purpose and site situation. Their validation requires comprehensive knowledge and data (Bellocchi et al., 2009). They run well in the environment they are created for, but their transferability to other locations, scales or purposes is limited. Their data input demand, effort for soil data adaptation to other environments, and their calculation time is currently relatively high as compared with straightforward soil and land rating approaches of Section 5.2.4. However, because of their sophisticated process-based background and further advances in technology, biomass prediction models have great potentials to serve as reliable and fast decision tools. Their flexibility in handling will remain limited in comparison with simple soil and land rating approaches.

#### 5.2.6 Direct Recordings of Biomass and Crop Yield Data

Crop yield is a part of the net primary production (NPP) in managed ecosystems. Yield and NPP are often satellite driven, recorded and modelled (Smit et al., 2008; Prieto-Blanco et al., 2009; Kurtz et al., 2009). Also, permanent recording of spatial crop yield data as done in precision farming (Ritter et al., 2008; Schellberg et al., 2008; Lukas et al., 2009) may produce databases which have the potential to predict the productivity of land by statistical procedures of spatiotemporal auto-regressive forecasting, state-space approaches (Wendroth et al., 2003) or combinations of models and data (Reuter et al., 2005; Schellberg et al., 2008). The latter approaches developed for precision farming may provide excellent GIS-based modelling or even forecasting of land productivity in the field and at a regional scale but algorithms are rarely transferrable to other regions. Over larger regions and at a range of scales, the availability of soil survey information has to be taken into account. The combination of soil information systems with recorded crop yield data allows an identification of crop-yield relevant soil properties.

All these approaches represent major areas of soil scientific progress over the past 40 years (Mermut and Eswaran, 2001) but include two common risks of data gathering. First, the speed in developing algorithms and models often cannot keep pace with the rate of increase of available data. A second implication may be the loss of "ground adhesion", e.g. the difficulty of incorporating large amounts of data and sophisticated models into participatory approaches of decision support and in-situ decision procedures. Soil quality assessments for sustainable land use require straightforward tools, reliable but easy to implement into more complex decision models. Approaches based on simple soil functional classifications which are cross-validated with satellite and aerial data show great versatility for modelling policy scenarios (Baisden, 2006).

## 5.3 Comparison of Methods of Soil Evaluation Relevant to Soil Productivity

The comparability of soil productivity-related methods for assessing overall soil quality has been evaluated by different criteria including scale of validity, field method capability, reliability, relation to soil and climate data, plant suitability and others. Table 1 shows a list of criteria applied for the evaluation of the methods. For reasons of overview and readability of the table, only the rating values of a few distinct methods are provided. Values demonstrate that all existing methods have their merits and weakness regarding specific criteria. Figure 3 is an arbitrary similarity-dissimilarity plot by neighbourhood for evaluating systems of soil productivity potentials using a statistical procedure of multi-dimensional scaling (Procedure MDS, SPSS Inc., 1993). This plot is a computed map based on extending Table 1 by including more available methods and weightings of some criteria like performance over scales and correlations with crop yields. Wide separations indicate dissimilarities of methods. This procedure shows clear separation between traditional soil ratings (Storie Index Rating, German Soil Rating and dynamic visual assessments of soil quality (VSA)). The rating system of the former Soviet Union (Gavrilyuk, 1974) is similar to the Storie Index Rating. Crop models and the AEZ methodology are similar both in purpose and in results. They are located far from the centre as these procedures are not field methods of soil assessment and are mainly based on climate information.

Soil data sets (examples: minimum data set of Wienhold et al. (2004), or Cornell soil health test (Schindelbeck et al., 2008), also occupy isolated positions as, although they contain detailed soil information, they do not contain climate information and are based on laboratory analyses.

The soil management assessment framework of Andrews et al. (2004) would also be located in their vicinity. The Canadian Land Suitability Rating System (LSRS), and the Muencheberg Soil Quality Rating, (M-SQR) which include more crop yield relevant parameters (climate, soil structure) are in-between and closer to the centre. While rating procedures are

**Table 1** Evaluation criteria and scheme of some existing methods for assessing overall agricultural soil quality (evaluation numbers 0 = none/false/worse; 1 = low/few/slow; 2 = medium; 3 = high/manv/much/fast/good; 3 is always the best rating)

Criterion ↓	Storie index <sup>(1)</sup>	German BS <sup>(2)</sup>	AEZ <sup>(3)</sup>	VSA (4)	M-SQR <sup>(5)</sup>
Purpose of method					
Overall soil rating	3	3	0-1	0	3
Capability rating potential	3	0	1-2	0-1	3
Crop suitability rating	0	0	3	0-1	0-1
Tool for soil monitoring	0	0	1	3	2-3
Tool for soil management/extension	0	0	0	3	2–3
Tool for land use planning	2–3	2	3	1	3
Performance in spatial scales					
Field to regional level	3	3	0	3	3
Large regional to nation level	3	3	3	2-3	3
Trans-National	2	1	3	1–2	3
Indicator criteria					
Number of inherent SQ <sup>(6)</sup> indicators <sup>(a)</sup>	2	1	2	0	3
Number of dynamic SQ indicators	0	0	0	2	1
Climate inclusion	0	0	3	0	2
Interactions between indicators considered	0	0-1	2-3	0	0-1
Potential for assessing soil functions other than productivity	1–2	1	1	1	2–3
Further key criteria					
Simplicity in the field	3	3	0	2	2
Applicable without soil test kits	2–3	3	3	3	2-3
Speed of field rating <sup>(b)</sup>	2–3	3	0	3	2-3
Changes with soil depth included?	2	2	1	1	3
Correlation of scores with crop yields					
Field to regional level	2	2	0-1	1-2	2
Large regional to nation level	1–2	1	3	1–2	2
Trans-National	1	0-1	3	0	2

Abbreviations and references: (1) Storie index (Storie, 1933), (2) German BS (German Soil Rating, Rothkegel, 1950), (3) AEZ (Agro-ecological zoning, Fischer et al., 2002), (4) VSA (Visual Soil Assessment, Shepherd, 2000), (5) M-SQR (Muencheberg Soil Quality Rating, Fig. 2, Mueller et al., 2007) (6) SQ (Soil Quality).

<sup>a</sup> Number of indicators/criteria 1, few <5, 2 medium (5–15), 3 high >15.

<sup>b</sup> Time required for field rating (minutes per pedon/unit): 3 fast < 20, 2 medium 20–40, 1 slow >40, 0 no field method.

different, inputs are similar. M-SQR indicator ratings are expert based and validated with crop yield data from Germany, Russia and China.

## 6 Targets and Steps to Assessing the Soil Productivity Function in the 21st Century

All approaches for assessing mainly regional-specific and particular aspects of the soil potential for productivity have their eligibility and merits. However, in the resource-limited global world of the 21st century we need more precise instruments for monitoring and controlling the functionality of the soil resource by clearly defined but not only locally valid criteria. A global soil functional assessment and classification framework will enable creation of reliable indicators of farmland quality, consistently over spatial scales, for example a reliable agri-environmental indicator "High quality farmland" which is currently not available. Based on our analysis such a global assessment framework of the soil productivity function has to meet the following requirements:

- a monitoring, controlling and modelling tool of the functional status of the soil resource for crop productivity;
- precise in operation, based on indicators and thresholds of the most functionally relevant



**Fig. 3** Similarity plot of some soil productivity-relevant evaluation systems. Similarity is expressed by local neighbourhood. Axes are based on computed complex factors and have thus arbitrary meaning. Abbreviations and references: German BS = German Soil Rating (Rothkegel, 1950), Austrian BS = Austrian Soil Rating (Bodenaufnahmesysteme, 2001), Storie index (Storie, 1933), M-SQR (Muencheberg Soil Quality Rating, Fig. 2, Mueller et al., 2007), VSA (Visual Soil Assessment, Shepherd, 2000), LSRS Canada (Land Suitability Rating System, Agronomic Interpretations Working Group, 1995), AEZ (Agro-ecological zoning, Fischer et al., 2002)

parameters identified as soil moisture and temperature regimes, and textural and structural soil attributes;

- consistently applicable over different scales, from a field method to global overviews based on the soil map of the world;
- potential for suitability and capability classifications;
- straightforward for the use in extension and enabling participatory assessments;
- relevant to crop performance, with potential as a crop yield estimator and thus acceptable to farmers and other stakeholders;
- compatible with existing FAO soil classifications and capable of being integrated into new land evaluation frameworks of the 21st century (FAO, 2007).

Both the Canadian Land Suitability Rating System and the Muencheberg Soil Quality Rating meet the majority of these criteria. They contain information on climate and soil properties relevant to crop yield, and soil structure in particular. They have the potential for consistent ratings of the soil productivity function on a global scale but they need to be tested and evolved for this purpose in major agricultural regions. The selection and quantification of indicators and definition of thresholds and testing of the accuracy and sensitivity of the overall rating outputs under different environments will be a task of high priority. The latest results of Huber et al. (2008) about identified indicators and thresholds for main threats and degradation risks of soils in the EU will also need to be integrated.

Recent calls and approaches for the standardisation of soil quality attributes and their analyses (Nortcliff, 2002; FAO, 2007; Schindelbeck et al., 2008) will be very important for comparing productivity relevant soil states over the globe. The selection of attributes, data sets and indicators is the basic problem, and needs also to be relevant on a global perspective. Further locally proven and tested approaches and their indicator sets and thresholds (Kundler, 1989; Wienhold et al., 2004; Zhang et al., 2004; Barrios et al., 2006; Ochola et al., 2006; Govaerts et al., 2006; Sparling et al., 2008) referring to typical regions or countries have to be tested on inclusion into the frameworks.

Key indicators are single highly relevant attributes reflecting complex systems. Besides soil structure, soil organic carbon is such a key indicator of soil quality, associated with many soil functions other than productivity. It is also beneficial to agricultural productivity (Kundler, 1989; Rogasik et al., 2001; Lal, 2006; Martin-Rueda et al., 2007; Pan et al., 2009; Jones et al., 2009) at a limited level of inputs of farming but specific targets or thresholds are difficult to specify (Sparling et al., 2003). Despite this difficulty, from a broader perspective of soil functionality, organic carbon must be evolved as a globally key indicator of agricultural soil quality.

#### 7 Conclusions

(i) There is a lack of a standardised methodology to assess soil productivity potentials for a growing global community of stakeholders achieving a sustainable use of the soil resource. Existing soil and land evaluation and classification systems operate on a regional or national basis. The soil types or reference groups of many existing soil classifications including the latest World Reference Base for Soil Resources are largely based on pedogenic criteria and provide insufficient information on soil functionality. A common internationally applicable method providing field soil productivity ratings is required but does not exist.

- (ii) We advocate a straightforward indicator-based soil functional evaluation and classification system supplementing the WRB soil classifications. This could provide a useful tool for monitoring and controlling the soil status for sustainable land use at an internationally comparable scale. It could also serve as a soil productivity estimator providing a fast appraisal of attainable crop yields over different scales.
- (iii) This framework has to meet the following criteria: precise in operation, based on indicators and thresholds of soil, consistently applicable over different scales, potential for suitability and capability classifications, adequately crop yield relevant, and capable of being integrated into new land evaluation frameworks of the 21st century.
- (iv) Evolving this framework based on favoured methods for this purpose, the Muencheberg Soil Quality Rating (M-SQR) and the Canadian Land Suitability Rating System (LSRS), will be a starting point for assessing sustainable agricultural productivity without compromising soil quality.

Acknowledgements Authors thank Dr. Eric Lichtfouse and two anonymous reviewers for their helpful suggestions and comments.

#### References

- AG Boden (2005) Bodenkundliche Kartieranleitung (KA5), 5th edition, Hannover, 432 p.
- Agronomic Interpretations Working Group (1995) Land Suitability Rating System for Agricultural Crops. 1. Springseeded small grains, in: Pettapiece W.W. (Ed.), Tech. Bull. 1995-6E, Centre for Land and Biological Resources Research, Agriculture and Agri-Food Canada, Ottawa, 90 p.
- Ahrens J.R., Rice T.J., Eswaran H. (2002) Soil Classification: Past and Present, NCSS Newslett. 19, 1–5.
- Alcamo J., Dronin N., Endejan M., Golubev G., Kirilenko A. (2007) A new assessment of climate change impacts on food production shortfalls and water availability in Russia, Global Environ. Change 17, 429–444.
- Andrews S.S., Karlen D.L., Cambardella C.A. (2004) The Soil Management Assessment Framework: A Quantitative Soil Quality Evaluation Method, Soil Sci. Soc. Am. J. 68, 1945–1962.

- Asner G.P., Heidebrecht K.B. (2005) Desertification alters regional ecosystem-climate Inter-actions, Glob. Change Biol. 11, 182–194.
- Bacic I.L.Z., Rossiter D.G., Bregt A.K. (2003) The use of land evaluation information by land use planners and decisionmakers: a case study in Santa Catarina, Brazil, Soil Use Manage. 19, 12–18.
- Baisden W.T. (2006) Agricultural and forest productivity for modelling policy scenarios: evaluating approaches for New Zealand greenhouse gas mitigation, J. Roy. Soc. New Zeal. 36, 1–15.
- Ball B.C., Batey T., Munkholm L.J. (2007) Field assessment of soil structural quality - a development of the Peerlkamp test, Soil Use Manage. 23, 329–337.
- Barrios E., Delve R.J., Bekunda M., Mowo J., Agunda J., Ramisch J. (2006) Indicators of soil quality: A South–South development of a methodological guide for linking local and technical knowledge, Geoderma 135, 248–259.
- Batey T., McKenzie D.C. (2006) Soil compaction: identification directly in the field, Soil Use Manage. 22, 123–131.
- BBodSchG (1998) Federal Ministry for the Environment, Nature Conservation and Nuclear Safety, Federal Soil Protection Act of 17 March 1998, Federal Law Gazette I, p. 502.
- Begon M.R., Townsend R., Harper J.R. (2006) Ecology: from individuals to ecosystems, 4th ed., Blackwell.
- Beinroth F.H., Stahr K. (2005) Geschichte und Prinzipien der Bodenklassifikation, in: Blume H.-P., Felix-Henningsen P., Fischer W., Frede H.-G., Guggenberger G., Horn, R., Stahr K. (Eds.), Handbuch der Bodenkunde, Ecomed. 23. Erg. Lfg. 11/05, Section 3.2.1., 22 p.
- Bellocchi G., Rivington M., Donatelli M., Matthews K. (2009) Validation of biophysical models: issues and methodologies. A review, Agron. Sustain. Dev. 29, 1–22.
- Bibby J.S., Douglas H.A., Thomasson A.J., Robertson J.S. (1991) Land capability classification for agriculture, Macaulay Land Use Research Institute, Aberdeen.
- Blum W.E.H. (1993) Soil Protection Concept of the Council of Europe and Integrated Soil Research, in: Eijsackers H.J.P., Hamer T. (Eds.), Integrated Soil and Sediment Research: A basis for Proper Protection, Soil and Environment, Dordrecht: Kluwer Academic Publishers, Vol. 1, pp. 37–47.
- Blum W.E.H. (2006) Soil Resources The basis of human society and the environment, Bodenkultur 57, 197–202.
- Bockstaller C., Guichard L., Keichinger O., Girardin P., Galan M.-B., Gaillard G. (2009) Comparison of methods to assess the sustainability of agricultural systems. A review, Agron. Sustain. Dev. 29, 223–235.
- Bodenaufnahmesysteme in Österreich (2001) Bodeninformationen für Land-, Forst-, Wasser- und Abfallwirtschaft, Naturschutz-, Landschafts-, Landes- und Raumplanung, Agrarstrukurelle Planung, Bodensanierung und -regeneration sowie Universitäten, Schulen und Bürger. Mitteilungen der Österreichischen Bodenkundlichen Gesellschaft Heft 62, zugleich eine Publikation des Umweltbundesamtes Wien, 2001, 221 p.
- Bondeau A., Smith C.M., Zaehle S., Schaphoff S., Lucht W., Cramer W., Gerten D., Lotze-Campen H., Mueller C., Reichstein M., Smith B. (2007) Modelling the role of agriculture for the 20th century global terrestrial carbon balance, Glob. Change Biol. 13, 679–706.

Borlaug N. (2007) Feeding a hungry world, Science 318, 359.

- Buol S.W., Sanchez P.A., Cate R.B., Granger M.A. (1975) Soil fertility capability classification: a technical soil classification system for fertility management, in: Bornemisza E., Alvarado A. (Eds.), Soil Management in Tropical America, N.C. State Univ., Raleigh, NC, pp. 126–145.
- Bronick C.J., Lal R. (2005) Soil structure and management: a review, Geoderma 124, 3–22.
- Brown I., Towers W., Rivington M., Black H.I.J. (2008) The influence of climate change on agricultural land-use potential: adapting and updating the land capability system for Scotland, Climate Research 37, 43–57.
- Cannell R.Q., Davies D.B., Mackney D., Pidgeon J.D. (1978) The suitability of soils for sequential direct drilling of combine-harvested crops in Britain: a provisional classification, Outlook Agric. 9, 306–316.
- Cassman K.G., Dobermann A., Walters D.T., Yang, H. (2003) Meeting Cereal Demand While Protecting Natural Resources and Improving Environmental Quality, Ann. Rev. Environ. Res. 28, 315–358.
- Ceotto E. (2008) Grasslands for bioenergy production. A review, Agron. Sustain. Dev. 28, 47–55.
- Ciais P., Reichstein M., Viovy N., Granier A., Ogée J., Allard V., Aubinet M., Buchmann N., Bernhofer C., Carrara A., Chevallier F., De Noblet N.A., Friend D., Friedlingstein P., Grünwald T., Heinesch B., Keronen P., Knohl A., Krinner G., Loustau D., Manca G., Matteucci G., Miglietta F., Ourcival J.M., Papale D., Pilegaard K., Rambal S., Seufert G., Soussana J.F., Sanz M.J., Schulze E.D., Vesala T., Valentini R. (2005) Europe-wide reduction in primary productivity caused by the heat and drought in 2003, Nature 437, 529–533.
- Davis M.L., Masten S.J. (2003) Principles of Environmental Engineering and Science, McGraw-Hill Professional, ISBN 0072921862, 9780072921861, 704 p.
- Debaeke P., Aboudrare A. (2004) Adaptation of crop management to water-limited environments, Eur. J. Agron. 21, 433–446.
- De la Rosa, D. (2005) Soil quality evaluation and monitoring based on land evaluation, Land Degrad. Dev. 16, 551–559.
- De la Rosa D., Anaya-Romero M., Diaz-Pereira E., Heredia R., Shahbazi F. (2009) Soil-specific agro-ecological strategies for sustainable land use – A case study by using MicroLEIS DSS in Sevilla Province (Spain), Land Use Policy 26, 1055–1065.
- Del Grosso S.J., Mosier A.R., Parton W.J., Ojima D.S. (2005) DAYCENT model analysis of past and contemporary soil N<sub>2</sub>O and net greenhouse gas flux for major crops in the USA, Soil Tillage Res. 83, 9–24.
- Duan X.W., Xie Y., Feng Y.J., Yin S.Q. (2009) Study on the Method of Soil Productivity Assessment in Black Soil Region of Northeast China, Agric. Sci. China 8, 472–481.
- Durán Zuazo, V.H., Rodríguez Pleguezuelo C.R. (2008) Soilerosion and runoff prevention by plant covers. A review, Agron. Sustain. Dev. 28, 65–86.
- EC (2006) COM 2006/231 2006, Communication from the Commission to the Council, the European Parliament, the European Economic and Sicial Committee and the Committee of the Regions- Thematic Strategy for Soil Protection, Commission of the European Communities, Brussels, 22.9.2006.

- Eswaran H., Almaraz R., van den Berg E., Reich P. (1997) An assessment of the soil resources of Africa in relation to productivity, Geoderma 77, 1–18.
- FAO (1976) A framework for land evaluation, FAO, Rome, FAO Soils Bull. 32.
- FAO (2006) Guidelines for Soil Description (4th ed.), FAO, Rome, 95 p.
- FAO (2007) Land evaluation, Towards a revised framework, Land and water discussion paper 6, 107 p.
- Farooq M., Wahid A., Kobayashi N., Fujita D., Basra S.M.A. (2009) Plant drought stress: effects, mechanisms and management, Agron. Sustain. Dev. 29, 185–212.
- Feller C.L., Thuries L.J.-M., Manlay R.J., Robin P., Frossard E. (2003) "The principles of rational agriculture" by Albrecht Daniel Thaer (1752–1828), An approach to the sustainability of cropping systems at the beginning of the 19th century, J. Plant Nutr. Soil Sci. 166, 687–698.
- Fischer G., Sun L. (2001) Model based analysis of future landuse development in China, Agric. Ecosyst. Environ. 85, 163–176.
- Fischer G., van Velthuizen H., Shah M., Nachtergaele F. (2002) Global Agro-ecological Assessment for Agriculture in the 21st Century: Methodology and Results, International Institute for Applied Systems Analysis, Laxenburg, Austria, 154 p.
- Flach K.W. (1986) Modeling of soil productivity and related land classification, in: Siderius W. (Ed.), Land evaluation for land- use planning and conservation in I sloping areas. International Workshop, Enschede, The Netherlands, 17–21 December 1984, Publication 40, International Institute for Land Reclamation and Improvement/ILRI, Wageningen, The Netherlands.
- Foley J.A., de Fries R., Asner G.P., Barford C., Bonan G., Carpenter S.R., Chapin F.S., Coe M.T., Daily G.C., Gibbs H.K., Helkowski J.H., Holloway T., Howard E.A., Kucharik C.J., Monfreda C., Patz J.A., Prentice I.C., Ramankutty N., Snyder P.K. (2005) Global consequences of land use, Science 309, 570–574.
- Franko U., Oelschlägel B., Schenk S. (1995) Simulation of temperature-, water- and nitrogen dynamics using the model CANDY, Ecol. Model. 81, 213–222.
- Gavrilyuk F.Y. (1974) Bonitirovka pochv, Moskva, Vysshaya shkola, 270 p.
- Govaerts B., Sayre K.D., Deckers J. (2006) A minimum data set for soil quality assessment of wheat and maize cropping in the highlands of Mexico, Soil Tillage Res. 87, 163–174.
- Hall R. (2008) Soil Essentials, Managing Your Farm's Primary Asset, Landlinks Press, 1st ed., 192 p.
- Hansen S., Jensen H.E., Nielsen N.E., Svendsen H. (1990) DAISY: Soil Plant Atmoshere System Model, NPO Report No. A10, The National Agency for Environmental Protection, Copenhagen, 272 p.
- Harrach T. (1982) Ertragsfähigkeit erodierter Böden, Arbeiten der DLG, Bd. 174, Bodenerosion, 84–91.
- Hartmann K.-J., Finnern J., Cordsen E. (1999) Bewertung von Bodenfunktionen auf Grundlage der Bodenschätzung, ein Verfahrensvergleich, J. Plant Nutr. Soil Sci. 162, 179–181.
- Helming K., Tscherning K., König B., Sieber S., Wiggering H., Kuhlman T., Wascher D., Perez-Soba M., Smeets P., Tabbush P., Dilly O., Hüttl R.F., Bach H. (2008) Ex ante impact assessment of land use change in European regions:

the SENSOR approach, in: Helming K., Pérez-Soba M., Tabbush P. (Eds.), Sustainability impact assessment of land use changes, Berlin, Springer, pp. 77–105.

- Helms D. (1992) Readings in the History of the Soil Conservation Service, Washington, DC, Soil Conservation Service, pp. 60–73.
- Hijmans R.J., Giuking-Lens I.M., van Diepen C.A. (1994) Usefs guide for the WOFOST 6.0 crop growth simulation model, Technical Document 12, DLO Winand Staring Centre, Wageningen, 145 p.
- Hillel D. (2009) The mission of soil science in a changing world, J. Plant Nutr. Soil Sci. 172, 5–9.
- Hillel D., Rosenzweig C. (2002) Desertification in relation to climate variability and change, Adv. Agron. 77, 1–38.
- Huber S., Prokop G., Arrouays D., Banko G., Bispo A., Jones R.J.A., Kibblewhite M.G., Lexer W., Möller A., Rickson R.J., Shishkov T., Stephens M., Toth G., Van den Akker J.J.H., Varallyay G., Verheijen F.G.A., Jones A.R. (2008) Environmental Assessment of Soil for Monitoring: Volume I, Indicators & Criteria. EUR 23490 EN/1, Office for the Official Publications of the European Communities, Luxembourg, 339 p.
- Hulugalle N.R., Weaver T.B., Finlay L.A., Hare J., Entwistle P.C. (2007) Soil properties and crop yields in a dryland Vertisol sown with cotton-based crop rotations, Soil Tillage Res. 93, 356–369.
- Jackson L.E., Calderon F.J., Steenwerth K.L., Scow K.M., Rolston D.E. (2003) Responses of soil microbial processes and community structure to tillage events and implications for soil quality, Geoderma 114, 305–317.
- Jones A., Stolbovoy V., Rusco E., Gentile A.-R., Gardi C., Marechal B., Montanarella L., (2009) Climate change in Europe. 2. Impact on soil. A review, Agron. Sustain. Dev. 29, 423–432.
- Karlen D.L. (2004) Soil quality as an indicator of sustainable tillage practices, Soil Tillage Res. 78, 129–130.
- Karlen D.L., Mausbach M.J., Doran J.W., Cline R.G., Harris R.F., Schuman G.E. (1997). Soil quality: a concept, definition and framework for evaluation, Soil Sci. Soc. Am. J. 61, 4–10.
- Karmanov I.I., Bulgakov D.S., Karmanova L.A., Putilin E.I. (2002) Modern aspects of the assessment of land quality and soil fertility, Eurasian Soil Sci. 35, 7, 754–760.
- Kavdir Y., Smucker A.J.M. (2005) Soil aggregate sequestration of cover crop root and shoot-derived nitrogen, Plant Soil 272, 263–276.
- Kay B.D., Hajabbasi M.A., Ying J., Tollenaar M. (2006) Optimum versus non-limiting water contents for root growth, biomass accumulation, gas exchange and the rate of development of maize (*Zea mays L.*), Soil Tillage Res. 88, 42–54.
- Kersebaum K.-C. (2007) Modelling nitrogen dynamics in soil– crop systems with HERMES, Nutr. Cycl. Agroecosys. 77, 39–52.
- Kersebaum K.-C., Wurbs A., Jong R. de, Campbell C.A., Yang J., Zentner R.P. (2008) Long-term simulation of soil– crop interactions in semiarid southwestern Saskatchewan, Canada, Eur. J. Agron. 29, 1–12.
- Keys to Soil Taxonomy (2006) Tenth Edition United States Department of Agriculture, Natural Resources Conservation Service, 333 p.

- Klingebiel A.A., Montgomery P.H. (1961) Land capability classification. USDA Agricultural Handbook 210, US Government Printing Office, Washington, DC.
- Kopp D., Schwanecke W. (2003) Standörtlich-naturräumliche Grundlagen ökologiegerechter Forstwirtschaft, Kessel Verlag, 2. ed., 262 p.
- Kundler P. (1989) Erhöhung der Bodenfruchtbarkeit, VEB Deutscher Landwirtschaftsverlag Berlin, 1st ed., 452 p.
- Kurtz D., Schellberg J., Braun M. (2009) Ground and Satellite Based Assessment of Rangeland Management in Sub-Tropical Argentina, Appl. Geogr. 29, in press.
- Lal R. (2006) Enhancing crop yield in the developing countries through restoration of soil organic carbon pool in agricultural lands, Land Degrad. Dev. 17, 187–209.
- Lal R. (2008) Soils and sustainable agriculture. A review, Agron. Sustain. Dev. 28, 57–64.
- Lal R. (2009) Soils and food sufficiency, A review, Agron. Sustain. Dev. 29, 113–133.
- Lavalle C., Micale F., Houston T.D., Camia A., Hiederer R., Lazar C., Conte C., Amatulli G., Genovese G. (2009) Climate change in Europe. 3. Impact on agriculture and forestry. A review, Agron. Sustain. Dev. 29, 433–446.
- Lichtfouse E., Navarrete M., Debaeke P., Souchère V., Alberola C. (2009) Sustainable Agriculture, Springer, 1st ed., 645 p., ISBN: 978-90-481-2665-1.
- Lieth H. (1975) Modeling the primary productivity of the world, in: Lieth H., Whittaker R.H. (Eds.), Primary Productivity of the Biosphere, Springer, Berlin, pp. 237–263.
- Lin H., Bouma J., Wilding L.P., Richardson J.L., Kutilek M., Nielsen D.R. (2005) Advances in hydropedology, in: Sparks D.L. (Ed.), Adv. Agron. 85, 2–76.
- Louwagie G., Gay S.H., Burrell A. (2009) Addressing soil degradation in EU agriculture: relevant processes, practices and policies, Report on the project 'Sustainable Agriculture and Soil Conservation (SoCo)', JRC Scientific and Technical Reports, ISSN 1018 5593, 209 p.
- Lukas V., Neudert L., Kren J. (2009) Mapping of soil conditions in precision agriculture, Acta Agrophys. 13, 393–405.
- Lynn I., Manderson A., Page M., Harmsworth G., Eyles G., Douglas G., Mackay A., Newsome P. (2009) Land Use Capability Survey Handbook - a New Zealand handbook for the classification of land, 3rd ed., 164 p.
- Martin-Rueda I., Munoz-Guerra L.M., Yunta F., Esteban E., Tenorio J.L., Lucena J.J. (2007) Tillage and crop rotation effects on barley yield and soil nutrients on a Calciortidic Haploxeralf, Soil Tillage Res. 92, 1–9.
- Mausel P.W., Carmer S.G., Runge E.C.A. (1975) Soil productivity indexes for Illinois counties and soil associations. Bulletin 752, University of Illinois at Urbana-Champaign, College of Agriculture, Agricultural Experiment Station, 51 p.
- McKenzie D.C. (2001) Rapid assessment of soil compaction damage. I. The SOILpak score, a semi-quantitative measure of soil structural form, Aust. J. Soil Res. 39, 117–125.
- Medvedev V.V., Bulygin S. Yu, Laktionova T.N., Derevyanko R.G. (2002) Criteria for the Evaluation of Ukrainian Land Suitability for Growing Cereal Crops, Eurasian Soil Sci. 35, 192–202 (Pochvovedenie, 2002, No. 2, 216–227).
- Mermut A.R., Eswaran, H. (2001) Some major developments of soil science since the mid-1960s, Geoderma 100, 403–426.
- Meyers Lexikon (1925) Bodenbonitierung, Meyers Lexikon, Bibliographisches Institut Leipzig, 7. Auflage, 2. Band, p. 567.
- Mirschel, W., Wenkel, K.-O. (2007) Modelling soil-crop interactions with AGROSIM model family, in: Kersebaum K.-C., Hecker J.-M., Mirschel W., Wegehenkel M. (Eds.), Modelling water and nutrient dynamics in soil crop systems: Proceedings of the workshop on "Modelling water and nutrient dynamics in soil-crop systems" held on 14–16 June 2004 in Müncheberg, Germany, Dordrecht Springer, pp. 59–73.
- Mueller L., Kay B.D., Hu C., Li Y., Schindler U., Behrendt A., Shepherd T.G., Ball B.C. (2009) Visual assessment of soil structure: Evaluation of methodologies on sites in Canada, China and Germany: Part I: Comparing visual methods and linking them with soil physical data and grain yield of cereals, Soil Tillage Res. 103, 178–187.
- Mueller L., Schindler U., Behrendt A., Eulenstein F., Dannowski R. (2007) Das Muencheberger Soil Quality Rating (SQR): ein einfaches Verfahren zur Bewertung der Eignung von Boeden als Farmland, Mitteilungen der Deutschen Bodenkundlichen Gesellschaft 110, 515–516.
- Mulengera M.K., Payton R.W. (1999) Modification of the productivity index model, Soil Tillage Res. 52, 11–19.
- Munkholm L.J., Schjonning P., Rasmussen K.J., Tanderup K. (2003) Spatial and temporal effects of direct drilling on soil structure in the seedling environment, Soil Tillage Res. 71, 163–173.
- Murray W.G., Harris D.G., Miller G.A., Thompson N.S. (1983) Farm appraisal and valuation, Iowa State University Press, 6th ed., 304 p.
- Nortcliff S. (2002) Standardisation of soil quality attributes, Agric. Ecosyst. Environ. 88, 161–168.
- Ochola W.D., Mwonya R., Mwarasomba L.I., Wambua M.M. (2006) Farm-level Indicators of Sustainable Agriculture, Classification and description of farm recommendation units for extension impact assessment in Koru, Kenya, in: Häni F.J., Pintér L., Herrens H.R. (Eds.), From Common Principles to Common Practice, Proceedings and outputs of the first Symposium of the International Forum on Assessing Sustainability in Agriculture (INFASA), March 16, 2006, Bern, Switzerland, pp. 49–76.
- Pan G., Smith P., Weinan Pan W. (2009) The role of soil organic matter in maintaining the productivity and yield stability of cereals in China, Agric. Ecosyst. Environ. 129, 344–348.
- Pease J., Coughlin R. (1996) Land Evaluation and Site Assessment: A Guidebook for Rating Agricultural Lands, Second Edition, prepared for the USDA Natural Resources Conservation Service, Soil and Water Conservation Society, Ankeny, IA.
- Peerlkamp P.K. (1967) Visual estimation of soil structure, in: de Boodt M., de Leenherr D.E., Frese H., Low A.J., Peerlkamp P.K. (Eds.), West European Methods for Soil Structure Determination, Vol. 2 (11), State Faculty Agricultural Science, Ghent, Belgium, pp. 216–223.
- Pehamberger A. (1992) Die Bodenschätzung in Österreich, Mitteilungen der Deutschen Bodenkundlichen Gesellschaft 67, S. 235–240.
- Peng L., Zhanbin L., Zhong Z. (2002) An Index System and Method for Soil Productivity Evaluation on the Hillsides in the Loess Plateau, 12th ISCO Conference Beijing 2002, Proceedings, pp. 330–342.

- Pierce F.J., Larson W.E., Dowdy R.H., Graham W.A.P. (1983) Productivity of soils: assessing long term changes due to erosion's long term effects, J. Soil Water Conserv. 38, 39–44.
- Poluektov R.A., Fintushal S.M., Oparina I.V., Shatskikh D.V., Terleev V.V., Zakharova E.T. (2002) AGROTOOL – A system for crop simulation, Arch. Agron. Soil Sci. 48, 609–635.
- Preetz H. (2003) Bewertung von Bodenfunktionen f
  ür die praktische Umsetzung des Boden-schutzes (dargestellt am Beispiel eines Untersuchungsgebiets in Sachsen-Anhalt), Ph-D thesis Halle-Wittenberg, 196 p.
- Prieto-Blanco A., North P.R.J., Barnsley M.J., Fox N. (2009) Satellite-driven modelling of Net Primary Productivity (NPP): Theoretical analysis, Remote Sens. Environ. 113, 137–147.
- Rao N.H., Rogers P.P. (2006) Assessment of agricultural sustainability, Curr. Sci. 91, 43–448.
- Reidsma P., Ewert F., Boogaard H., v. Diepen K. (2009) Regional crop modelling in Europe: The impact of climatic conditions and farm characteristics on maize yields, Agric. Syst. 100, 51–60.
- Reuter H.I., Kersebaum K.-C., Wendroth O. (2005) Modelling of solar radiation influenced by topographic shading: evaluation and application for precision farming, Phys. Chem. Earth 30, 143–149.
- Ritchie J.T., Godwin D.C. (1993) Simulation of Nitrogen Dynamics in the Soil Plant System with the CERES-models, Agrarinformatik 24, 215–230.
- Ritter C., Dicke D., Weis M., Oebel H., Piepho H.P., Büchse A., Gerhards R. (2008) An on-farm approach to quantify yield variation and to derive decision rules for site-specific weed management, Precis. Agric. 9, 133–146.
- Rogasik J., Schroetter S., Schnug E., Kundler P. (2001) Langzeiteffekte ackerbaulicher Massnahmen auf die Bodenfruchtbarkeit, Arch. Agron. Soil Sci. 47, 3–17.
- Roger-Estrade J., Richard G., Caneill J., Boizard H., Coquet Y., De' Fossez P., Manichon H. (2004) Morphological characterisation of soil structure in tilled fields. From a diagnosis method to the modelling of structural changes over time, Soil Tillage Res. 79, 33–49.
- Roger-Estrade J., Richard G., Dexter A.R., Boizard H., De Tourdonnet S., Bertrand M., Caneill J. (2009) Integration of soil structure variations with time and space into models for crop management. A review, Agron. Sustain. Dev. 29, 135–142.
- Rossiter D.G. (1996) A theoretical framework for land evaluation, Geoderma 72, 165–202.
- Rothkegel W. (1950) Geschichtliche Entwicklung der Bodenbonitierungen und Wesen und Bedeutung der deutschen Bodenschätzung, Stuttgart, Ulmer, 147 p.
- Rust I. (2006) Aktualisierung der Bodenschätzung unter Berücksichtigung klimatischer Bedingungen, Ph-D Thesis, Göttingen, 281 p.
- Sanchez P.A., Couto W., Buol S.W. (1982) The fertility capability soil classification system: Interpretation, application and modification, Geoderma 27, 283–309.
- Sanchez P.A., Palm C.A., Buol S.W. (2003) Fertility capability soil classification: a tool to help assess soil quality in the tropics, Geoderma 114, 157–185.

- Schellberg J., Hill M., Gerhards R., Rothmund M., Braun M. (2008) Precision agriculture on grassland: applications, perspectives and constraints - a review, Eur. J. Agron. 29, 59–71.
- Scherr S.J. (1999) Soil Degradation. A Threat to Developing-Country Food Security by 2020? Food, Agriculture, and the Environment Discussion Paper 27, International Food Policy Research Institute, Washington, DC 20006-1002, USA.
- Schindelbeck R.S., van Es H.M., Abawi G.S., Wolfe D.W., Whitlow T.L., Gugino B.K., Idowu O.J., Moebius-Clune B.N. (2008) Comprehensive assessment of soil quality for landscape and urban management, Landsc. Urban Plan. 88, 73–80.
- Schroeder D., Lamp J. (1976) Prinzipien der Aufstellung von Bodenklassifikationssystemen, Z. Pflanzenernaehr. Bodenk. 5, 617–630.
- Shaxson T.S. (2006) Re-thinking the conservation of carbon, water and soil: a different perspective, Agron. Sustain. Dev. 26, 9–19.
- Shepherd T.G. (2000) Visual soil assessment, Volume 1, Field guide for cropping and pastoral grazing on flat to rolling country, Horizons.mw/Landcare Research, Palmerston North, 84 p.
- Shepherd T.G. (2009) Visual Soil Assessment. Volume 1. Field guide for pastoral grazing and cropping on flat to rolling country, 2nd ed., Horizons Regional Council, Palmerston North, New Zealand, 118 p.
- Smit H.J., Metzger M.J., Ewert F. (2008) Spatial distribution of grassland productivity and land use in Europe, Agric. Syst. 98, 208–219.
- Smith P., Martino D., Cai Z., Gwary D., Janzen H., Kumar P., McCarl B., Ogle S., O'Mara F., Rice C., Scholes B., Sirotenko O. (2007) Agriculture, In Climate Change 2007: Mitigation. Contribution of Working Group III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, in: Metz B., Davidson O.R., Bosch P.R., Dave R., Meyer L.A. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Sparling G., Lilburne L., Vojvodić-Vuković M. (2008) Provisional Targets for Soil Quality Indicators in New Zealand, Landcare Research Science Series No. 34, Lincoln, Canterbury, New Zealand.
- Sparling G., Parfitt R.L., Hewitt A.E., Schipper L.A. (2003) Three Approaches to Define Desired Soil Organic Matter Contents, J. Environ. Qual. 32, 760–766.
- SPSS Inc. (1993) Handbooks SPSS for Windows, Release 6.0, Advanced statistics, 578 p., Professional statistics, 385 p.
- Stehfest E., Heistermann M., Priess J.A., Ojima D.S., Alcamo J. (2007) Simulation of global crop production with the ecosystem model DayCent, Ecol. Model. 209, 203–219.
- Stenitzer E., Murer E. (2003) Impact of soil compaction upon soil water balance and maize yield estimated by the SIMWASER model, Soil Tillage Res. 73, 43–56.
- Storie R.E. (1933) An index for rating the agricultural value of soils, Agricultural Experiment, Station Bulletin 556, University of California Agricultural Experiment Station, Berkley, CA.
- Supit I., Hooijer A.A., van Diepen C.A. (1994) EUR 15956 - System description of the WOFOST 6.0 crop simulation model implemented in CGMS (Volume 1: Theory and Algorithms), European Commission,

Luxembourg: Office for Official Publications of the European Communities, Agricultural series, Catalogue number: CL-NA-15956-EN-C, 146 p.

- Tan G.X., Shibasaki R. (2003) Global estimation of crop productivity and the impacts of global warming by GIS and EPIC integration, Ecol. Model. 168, 357–370.
- Tang H., van Ranst E., Sys C. (1992) An Approach to Predict Land Production Potential for Irrigated and Rainfed Winter Wheat in Pinan County, China, Soil Technol. 5, 213–224.
- Tóth G., Stolbovoy V., Montanarella L. (2007) Institute for Environment and Sustainability, Soil quality and sustainability evaluation, An integrated approach to support soil-related policies of the European Union, A JRC position paper, 40 p.
- Tóth T., Pásztor L., Várallyay G., Tóth G. (2007) Overview of soil information and soil protection policies in Hungary, in: Hengl T., Panagos P., Jones A., Tóth G. (Eds.), Status and prospect of soil information in southeastern europe: soil databases, projects and applications, Institute for Environment and Sustainability, 189 p., pp. 77–86.
- Van de Steeg J. (2003) Land evaluation for agrarian reform. A case study for Brasil, Landbauforschung Völkenrode, FAL Agricultural Research, Special Issue No. 246, 108 p.
- Van Diepen C.A., van Keulen H., Wolf J., Berkhout J.A.A. (1991) Land evaluation: From intuition to quantification, in: Stewart B.A. (Ed.), New York: Springer, Adv. Soil Sci. 15, 139–204.
- Verdoodt A., van Ranst E. (2006) Environmental assessment tools for multi-scale land resources information systems. A case study of Rwanda, Agric. Ecosyst. Environ. 114, 170–184.
- Verhulst N., Govaerts B., Sayre K.D., Deckers J., François I.M., Dendooven L. (2009) Using NDVI and soil quality analysis to assess influence of agronomic management on within-plot spatial variability and factors limiting production, Plant Soil 317, 41–59.
- Walter C., Stützel H. (2009) A new method for assessing the sustainability of land-use systems (I): Identifying the relevant issues, Ecol. Econ. 68, 1275–1287.
- Wegehenkel M., Mirschel W., Wenkel K.-O. (2004) Predictions of soil water and crop growth dynamics using the agroecosystem models THESEUS and OPUS, J. Plant Nutr. Soil Sci. 167, 736–744.
- Wendroth O., Reuter H.I., Kersebaum K.C. (2003) Predicting yield of barley across a landscape: a state-space modeling approach, J. Hydrol. 272, 250–263.
- Wienhold B.J., Andrews S.S., Karlen D.L. (2004) Soil quality: a review of the science and experiences in the USA, Environ. Geochem. Health 26, 89–95.
- Wiggering H., Dalchow C., Glemnitz M., Helming K., Mueller K., Schultz A., Stachow U., Zander P. (2006) Indicators for multifunctional land use: linking socio-economic requirements with landscape potentials, Ecol. Ind. 6, 238–249.
- Williams J.R., Dyke P.T., Jones C.A. (1983) EPIC a model for assessing the effects of erosion on soil productivity, in: The Third International Conference on State of the Art Ecological Modelling, Elsevier, Amsterdam, pp. 553–572.
- Wong M.T.F., Asseng S. (2006) Determining the causes of spatial and temporal variability of wheat yields at sub-field scale using a new method of upscaling a crop model, Plant Soil 283, 203–215.

- WRB (2006) World Reference Base for Soil Resources 2006, A Framework for International Classification, Correlation and Communication, FAO Rome, 2006, World Soil Resources Reports 103, 145 p.
- Xiong W., Conway D., Holman I., Lin E. (2008) Evaluation of CERES-Wheat simulation of Wheat Production in China, Agron. J. 100, 1720–1728.
- Yakovlev A.S., Loiko P.F., Sazonov N.V., Prokhorov A.N., Sapozhnikov P.M. (2006) Legal Aspects of Soil Conservation and Land Cadaster Works, Eurasian Soil Sci. 39, 693–698.
- Zhang B., Zhang Y., Chen D., White R.E., Li Y. (2004) A quantitative evaluation system of soil productivity for intensive agriculture in China, Geoderma 123, 319–331.

# Long-Term Effects of Organic Amendments on Soil Fertility

Mariangela Diacono and Francesco Montemurro

### Contents

1	Intro	duction	762
	1.1	Soil Management Strategies for a Sustainable	
		Agriculture	763
	1.2	Purpose of this Review Article	765
2	Effec	ts of Organic Amendments on the	
	Soil-	Plant System	765
	2.1	Effects on Soil Biological, Chemical and	
		Physical Fertility	768
	2.2	Effects of Organic Amendments on Plant	
		Nutrition and Yielding Responses	773
3	Envi	ronmental Impact and Sustainability	
	of Or	ganic Amendments	776
	3.1	Organic Matter Evolution and Soil Carbon	
		Sequestration	777
	3.2	Heavy Metal Concentration in the	
		Agro-Ecosystem	778
	33	N Pool Fate	780
	5.5		
4	Conc	lusion	782

CRA - Research Unit for the Study of Cropping Systems – Metaponto (MT), Italy e-mail: francesco.montemurro@entecra.it Abstract Common agricultural practices such as excessive use of agro-chemicals, deep tillage and luxury irrigation have degraded soils, polluted water resources and contaminated the atmosphere. There is increasing concern about interrelated environmental problems such as soil degradation, desertification, erosion, and accelerated greenhouse effects and climate change. The decline in organic matter content of many soils is becoming a major process of soil degradation, particularly in European semi-arid Mediterranean regions. Degraded soils are not fertile and thus cannot maintain sustainable production. At the same time, the production of urban and industrial organic waste materials is widespread. Therefore, strategies for recycling such organic waste in agriculture must be developed. Here, we review long-term experiments (3–60 years) on the effects of organic amendments used both for organic matter replenishment and to avoid the application of high levels of chemical fertilizers. The major points of our analysis are: (1) many effects, e.g. carbon sequestration in the soil and possible build-up of toxic elements, evolve slowly, so it is necessary to refer to long-term trials. (2) Repeated application of exogenous organic matter to cropland led to an improvement in soil biological functions. For instance, microbial biomass carbon increased by up to 100% using high-rate compost treatments, and enzymatic activity increased by 30% with sludge addition. (3) Long-lasting application of organic amendments increased organic carbon by up to 90% versus unfertilized soil, and up to 100% versus chemical fertilizer treatments. (4) Regular addition of organic residues, particularly the composted ones, increased soil physical fertility, mainly by improving aggregate stability and decreasing soil bulk density. (5) The best

F. Montemurro (🖂)

M. Diacono and F. Montemurro

agronomic performance of compost is often obtained with the highest rates and frequency of applications. Furthermore, applying these strategies, there were additional beneficial effects such as the slow release of nitrogen fertilizer. (6) Crop yield increased by up to 250% by long-term applications of high rates of municipal solid waste compost. Stabilized organic amendments do not reduce the crop yield quality, but improve it. (7) Organic amendments play a positive role in climate change mitigation by soil carbon sequestration, the size of which is dependent on their type, the rates and the frequency of application. (8) There is no tangible evidence demonstrating negative impacts of heavy metals applied to soil, particularly when high-quality compost was used for long periods. (9) Repeated application of composted materials enhances soil organic nitrogen content by up to 90%, storing it for mineralization in future cropping seasons, often without inducing nitrate leaching to groundwater.

**Keywords** Soil fertility • Soil carbon and organic matter • Organic amendments • Long-term experiments

### 1 Introduction

Indiscriminate use of agro-chemicals, excessive and deep tillage, and luxury irrigation have degraded soils, particularly in semi-arid Mediterranean areas, as well as polluted surface and groundwaters, and contaminated air (Lal, 2008). Soil is an essential nonrenewable resource with potentially rapid degradation rates and extremely slow formation and regeneration processes (Van-Camp et al., 2004). Soil degradation is as old as agriculture itself, its impact on human food production and the environment becoming more serious than ever before, because of its extent and intensity (Durán Zuazo and Rodríguez Pleguezuelo, 2008). There is a strong link between soil degradation and desertification on the one hand, and risks of accelerated greenhouse effects and climate change on the other (Komatsuzaki and Ohta, 2007). Durán Zuazo and Rodríguez Pleguezuelo (2008), confirmed that reduced precipitation or increased temperature accelerates land degradation through the loss of plant cover, biomass turnover, nutrient cycling and soil organic carbon storage, accompanied by higher greenhouse emissions. The reason for the earth's increased temperature along with change in rainfall amount and distribution, must be sought in the use of fossil fuel, that has drastically disturbed the global carbon cycle with its attendant impact on climate change (Lal, 2008). According to Komatsuzaki and Ohta (2007), there is also evidence that continuous cropping and inadequate replacement of nutrients, removed in harvested materials or lost through erosion, leaching, and gaseous emissions, degrade soil physical, chemical and biological properties, intensifying global warming. Moreover, the production of urban and industrial organic wastes is increasing worldwide, and strategies for disposal in such a way that these do not further degrade soil, contaminate water or pollute air must be developed and optimized (Düring and Gäth, 2002; Lal, 2008).

Soil fertility can be defined as the capacity of soil to provide physical, chemical and biological needs for the growth of plants for productivity, reproduction and quality, relevant to plant and soil type, land use and climatic conditions (Abbott and Murphy, 2007). It is becoming understandable that the proper agricultural use of soil resources requires equal consideration for biological, chemical and physical components of soil fertility, thus attaining a sustainable agricultural system.

The term "sustainable agriculture" is used in this review with the meaning given by Tilman et al. (2002), as referring to practices that meet current and future society needs for food and feed, ecosystem services and human health, maximizing the net benefit for people. Namely, sustainability implies both high yields, that can be maintained, and acceptable environmental impact of agricultural management.

It is relevant to note that organic farming is the only sustainable form of agriculture legally defined. According to the current legislation (EC Council Regulation N. 834, 2007), soil fertility management relies on a complex long-term integrated approach rather than the more short-term one of conventional agriculture. One of the possible main tools for the maintenance and the improvement of soil fertility in organic farming is to adopt crop rotations, including a mixture of leguminous fertility-building crops and plants with different rooting depths (Watson et al., 2002). Furthermore, organic wastes such as animal manures, by-products of several kinds and composted residues can be used as amendments to increase soil fertility, since they are important sources of nutrients for growing crops and means for enhancing the overall soil quality (Davies and Lennartsson, 2005).

## 1.1 Soil Management Strategies for a Sustainable Agriculture

Soil organic matter plays an important role in long-term soil conservation and/or restoration by sustaining its fertility, and hence in sustainable agricultural production, due to the improvement of physical, chemical and biological properties of soils (Sequi, 1989). The organic matter content is the result of the inputs by plant, animal and microbial residues, and the rate of decomposition through mineralization of both added and existing organic matter. More specifically, the generic term "organic matter" refers to the sum of all organic substances present in the soil. This sum comes from residues at various stages of decomposition, substances synthesized through microbialchemical reactions and biomass of soil microorganisms as well as other fauna, along with their metabolic products (Lal, 2007). Decomposition of organic matter is chiefly carried out by heterotrophic microorganisms. This process is under the influence of temperature, moisture and ambient soil conditions and leads to the release and cycling of plant nutrients, especially nitrogen (N), sulfur and phosphorus (Murphy et al., 2007).

The different fractions of organic matter undergo the "humification" process, which is the changing from recognizable parts and pieces of plants or animals into an amorphous, rotted dark mass. The products of humification are the humic substances that in soil are dark brown and fully decomposed, i.e. humified (Fig. 1). The humic substances are one of the most chemically active compounds in soils, with cation and anion exchange capacities far exceeding those of clays. They are long-lasting critical components of natural soil systems, persisting for hundreds or even thousands of years (Mayhew, 2004). In fact, the turnover rate of organic materials varies considerably, from less than 1 year, as for microbial biomass, to more than 1 thousand years of stable humus (Van-Camp et al., 2004). The organic matter is being progressively depleted, particularly in the Mediterranean area, where the warm climate and the intensity of cultivation increase the rate of decomposition (Montemurro et al., 2007). By contrast, the build-up of organic matter in soils is a process much slower and more complex than its decline (Van-Camp et al., 2004). Since organic matter contents are difficult to measure directly, a great number of methods measure the soil organic carbon level, multiplying it by conversion factors ranging from 1.7



**Fig. 1** The diagnostic horizons of a Vertisol (Sparacia experimental farm - Department of Agronomy, University of Palermo) showing the presence of well-transformed, dark-colored humi-fied organic matter in the topsoil

to 2.0 to obtain organic matter values (Baldock and Nelson, 2000).

Several investigations have demonstrated that soil organic matter is a very reactive and ubiquitous soil quality indicator that influences the productivity and physical well-being of soils (Lal, 2006; Komatsuzaki and Ohta, 2007). As a consequence, agricultural management practices that enhance soil organic matter content are used for preserving farming output and environmental quality; thereby they can be considered as sustainable activities (Lal, 2004).

Conservative soil tillage systems, e.g. no-till, which leaves more residues on the surface because the soil is not turned over, can maintain or improve the organic carbon content and the related soil fertility properties (Ismail et al., 1994; Johnson et al., 2005). Crop rotations usually increase organic matter and prompt changes in N sources, affecting their availability for plants and, as a consequence, the N efficiency is greater when a crop rotation is adopted (Montemurro and Maiorana, 2008).

The inclusion in a rotation of cover crops or green manures can also enhance the efficient use of nutrients by plants, mainly owing to the increase in soil microbial population and activity (Watson et al., 2002). Cover crops are generally grown to provide soil cover during the winter months, thus preventing soil erosion by wind and rainwater strength, which reduces organic matter content in the long run. Moreover, Komatsuzaki (2004) indicated that cover crop utilization is a technique that limits nutrient leaching, scavenging the soil residual N and making it available for subsequent cultivation.

In addition, the summer green manures are field crops or forage ones, such as leguminous and nonleguminous plants, usually incorporated into the soil soon after flowering, to improve soil fertility. In particular, leguminous green manures can fix large quantities of atmospheric N<sub>2</sub>. They also provide useful amounts of organic matter, as well as non-leguminous crops which, nevertheless, cannot fix atmospheric N<sub>2</sub> (Davies and Lennartsson, 2005). During 32 years of winter wheat cropping, Procházková et al. (2003) found higher average yields with green manuring compared with straw incorporation into the soil, probably due to a slower decomposition and following release of nutrients by the latter.

There is increasing interest in the alternative fertility building strategies previously described, because conventional inputs, such as synthetic fertilizers, should be excluded or reduced in sustainable agricultural management. Within this context, soil fertility could also be improved with organic waste application such as by-products of farming, or municipal activities including animal manures, food processing wastes and municipal biosolids, wastes from some industries, such as sewage sludges, wastewaters, husks and vinasse. Both groups of wastes present generally notable contents of organic matter and substantial quantities of nutrients and their use in agriculture can contribute to closing the natural ecological cycles (Montemurro et al., 2004; Montemurro and Maiorana, 2008). Increased recycling of organic residues as fertilizers and soil amendments on cropland avoids both utilization of non-renewable resources, e.g., fossil fuel and peat, and excess of energy expenses, i.e., production of chemical fertilizers and pesticides, treatment and landfill disposal of such organic wastes (Mondini and Sequi,

2008). Biodegradable wastes can also be considered valuable resources to promote soil fertility. However, this benefit occurred only if they were applied according to good practices, taking into account the needs of the soil, its use and the climatic conditions (Van-Camp et al., 2004). Moreover, it is necessary to adopt waste management strategies, such as controlled biodegradation processes, able to both minimize their potentially negative environmental impact and increase agricultural utilization (Montemurro et al., 2009).

Despite the fact that manure characteristics are influenced by many factors such as species and age of the animal, ration fed, and collection and storage method, it can be generally assumed that their ratio of nutrients are different from those removed by common crops (Edmeades, 2003). As a consequence, soil manure application, often in excess of crop requirements, can cause a significant build-up of phosphorus (P), N and salt. Nowadays, the industrialization of livestock enterprises has led to other problems linked to the land distribution of solid and liquid animal manures, enhancing the interest in pretreatment technologies, which can convert manure into a more valuable and easily usable resource called "compost" (Figs. 2a and 2b).

Compost is a stabilized and sanitized product of composting, which is the biodegradation process of a mixture of organic substrates carried out by a microbial community composed of various populations, both in aerobic conditions and solid state (Insam and de Bertoldi, 2007). During the composting process, the simple carbonaceous and nitrogenous compounds are transformed through the activity of microorganisms into more stable complex organic forms, which chemically resemble soil humic substances (Epstein, 1997).

The soil application of co-composted manure has several advantages over fresh manure, such as reduced numbers of viable weed seeds, reduced volume and particle size, which facilitates land distribution, a better balanced nutrient composition, stabilized organic matter and a slower release of nutrients. This topic has recently been reviewed by Moral et al. (2009).

From all the above, it can be summarized that the environmental impact of conventional farming practices and global concerns about soil degradation have increased the interest in sustainable agricultural strategies such as land application of waste materials. This is a way to avoid disposal costs and recycle nutrients into soil, unlike commercial fertilizers (Miller and



**Fig. 2** Aerobic biodegradation can enhance the quality of wastes which will be applied to soils. Figure 2a shows a composting windrow stirred mechanically by a turning machine, while Figure 2b shows an experimental field with compost application

Miller, 2000; Van-Camp et al., 2004). However, it is necessary to point out that the utilization of various organic residues in agriculture depends on several factors, including the characteristics of the waste, such as nutrient and toxic element content, availability, the transportation costs and the environmental regulations, as reviewed in detail by Westerman and Bicudo (2005).

#### 1.2 Purpose of this Review Article

The Rothamsted experiments, lasting for more than 100 years, are the oldest and most continuous agro-

nomic trials in the world, that measure the effects on crop yields of inorganic fertilizers in comparison with farmyard manure and other organic materials. Since their results are well summarized in the "Guide to the classical and other long-term experiments, datasets and sample archive", they will not be discussed further in this review (Rothamsted Research, 2006).

This paper, focusing on recently published data, gives emphasis to long-term field trials, particularly regarding raw and composted agro-industrial and municipal waste application. Although there is a large amount of literature relating to the influence of raw and composted organic materials on soil fertility, only a few published studies have focused on studies about long-term effects of these amendments on the soilplant system, for a sustainable crop production. The present work attempts to address this issue by using the results from long-lasting fertility trials. Long-term research on organic amendment use on cropland is particularly relevant because many effects, e.g. organic matter enrichment and possible soil toxic element accumulation, evolve slowly and are difficult to predict.

We have collected published literature, investigating a broad array of organic residues, composts and experimental conditions, especially analyzing their impact on soils and crops in relation to modern sustainable agriculture. Experimental data of field trials were selected lasting for at least 3 years up to 60 years, stressing longer-term research. Therefore, experimental designs considered in this review ranged from those usually known as mid- to long-term ones. Summaries of the essential data for the longest duration trials are given in Table 1.

The organic amendments covered in this paper are defined in Section 2 along with their effects on the soilplant system, whereas the environmental impact and sustainability are outlined in Section 3.

# 2 Effects of Organic Amendments on the Soil-Plant System

Miller and Miller (2000) highlighted that organic material application to cropland could affect soil properties, but the effects generally may not be apparent over a short time period. More specifically, Tittarelli et al. (2007) pointed out that the simplest method of examining the agronomic value of stabilized

Site	Organic materials	Application rate	Crop	Trial period	Reference
Punjab, India Obere Lobau near Vienna, Austria	Rice straw compost Biowaste compost	8 t ha <sup>-1</sup> 9, 16 and 23 t ha <sup>-1</sup>	Rice–wheat rotation Cereals and potatoes	10 years 10 years	Sodhi et al. (2009) Erhart et al. (2005); Hartl and Erhart (2005); Erhart et al. (2008)
Turin, Italy	<ol> <li>(1) Cattle slurry;</li> <li>(2) Composted farmyard manure</li> </ol>	<ul> <li>(1) 100 t ha<sup>-1</sup>;</li> <li>(2) 40 t ha<sup>-1</sup></li> </ul>	Maize for silage	11 years	Monaco et al. (2008)
Linz, Austria	Composts of: urban organic waste; green waste; cattle manure or sewage sludge	175 kg N ha <sup>-1</sup>	Maize, summer- wheat and winter-barley rotation	12 years	Ros et al. (2006a)
Bennett, CO, USA	Anaerobically digested biosolids	2.2, 4.5, 6.7, 8.9 and 11.2 t ha <sup>-1</sup>	Two-year wheat– fallow rotation	12 years	Barbarick and Ippolito (2007)
Ravenna, Italy	<ul> <li>Municipal-industrial wastewater sludge:</li> <li>(1) Anaerobically digested (liquid slurry);</li> <li>(2) belt filtered (dewatered sludge);</li> <li>(3) composted with wheat straw</li> </ul>	(1)–(3) 5 and 10 t ha <sup>-1</sup>	Winter wheat– maize–sugarbeet rotation	12 years	Mantovi et al. (2005)
Toledo, Spain	<ol> <li>Barley straw and crop waste;</li> <li>two-year-old cattle manure</li> </ol>	<ol> <li>(1) 3 and 2.5 t ha<sup>-1</sup>;</li> <li>(2) 30 t ha<sup>-1</sup></li> </ol>	Barley, wheat and sorghum	16 years	Dorado et al. (2003)
Murcia, Spain	Municipal solid waste	65, 130, 195 and 260 t ha <sup><math>-1</math></sup>	none	17 years	Bastida et al. (2008)
Orange, VA, USA	Aerobically digested sewage sludge	42, 84, 126,168 and 210 t ha <sup>-1</sup>	Barley, radish and romaine lettuce	19 years	Sukkariyah et al. (2005)
Tanashi City, Tokyo, Japan	<ol> <li>Dried sewage sludge;</li> <li>rice husk compost;</li> <li>sawdust compost</li> </ol>	<ol> <li>(1) 259.8 t ha<sup>-1</sup>;</li> <li>(2) 82.5 t ha<sup>-1</sup>;</li> <li>(3) 77.2 t ha<sup>-1</sup></li> </ol>	Maize, barley and rye	19 years	Kunito et al. (2001)
Tokyo, Japan	<ol> <li>Sewage sludge composted with rice husk;</li> <li>sewage sludge composted with sawdust</li> </ol>	(1)–(2) 240 kg N ha <sup>-1</sup>	Maize and barley	23 years	Zaman et al. (2004)
Alberta, Canada	Solid beef cattle manure	30, 60 and 90 t ha <sup><math>-1</math></sup> in dryland soils and 60, 120 and 180 t ha <sup><math>-1</math></sup> in irrigated soils	Barley, canola, triticale and maize	25 years	Whalen and Chang (2002)

**Table 1** Summary of the main data of long-term trials (selected data from experiments of  $\ge 10$  years)

(continued)

Site	Organic materials	Application rate	Crop	Trial period	Reference
Yamaguchi, Japan	Rice straw-cow dung compost	15 t ha <sup>-1</sup>	Double cropping (paddy rice and barley)	25 years	Shindo et al. (2006)
Czech Republic	<ol> <li>Straw harvest;</li> <li>straw harvest + green manuring;</li> <li>straw incorporation;</li> <li>straw incorporation + green manuring;</li> <li>straw burning;</li> <li>farmyard manure</li> </ol>	(6) 10 t ha <sup>-1</sup>	Winter wheat continuous cropping	32 years	Procházková et al. (2003)
Punjab, India	Farmyard manure	20 t ha <sup>-1</sup>	Rice–wheat and maize–wheat systems	34 years	Kukal et al. (2009)
Gumpenstein, Austria	<ol> <li>Cattle slurry + straw;</li> <li>animal manure (solid);</li> <li>animal manure (liquid);</li> <li>cattle slurry (semi-liquid)</li> </ol>	(1) 120 kg N ha <sup>-1</sup> + 0.2 kg m <sup>-2</sup> ; (2)–(4) 240 kg N ha <sup>-1</sup>	Cereals; rape; pea; flax	38 years	Antil et al. (2005)
Fidenza, Italy	<ol> <li>Farmyard manure;</li> <li>aerobically digested sewage sludge</li> </ol>	(1)–(2) 5 t ha <sup>-1</sup> digested sewage maize and wheat	Two-year rotation consisting mainly of	<ol> <li>(1) 40 years;</li> <li>(2) 12 years</li> </ol>	Saviozzi et al. (1999)
Legnaro, Italy	<ol> <li>(1) Farmyard manure;</li> <li>(2) liquid manure</li> </ol>	(1) 20 and 60 t ha <sup>-1</sup> ; (2) 120 t ha <sup>-1</sup>	Maize	40 years	Nardi et al. (2004)
Bologna, Italy	Cattle manure, cattle slurry, wheat or corn residues	6 t ha <sup>-1</sup> after wheat and 7.5 t ha <sup>-1</sup> after maize crops	Two-year maize- winter wheat rotation	34 years	Triberti et al. (2008)
Central Sweden	Farmyard manure	$4 t \text{ carbon ha}^{-1}$	Maize	47 years	Elfstrand et al. (2007)
Martonvásár and Keszthely Hungary	Farmyard manure	35, 40, 80 and 104 t $ha^{-1}$	Martonvásár: wheat–wheat– maize–maize; Keszthely:wheat– wheat–sugar beet–maize–maize	Martonvásár: 54 years; Keszthely: 41 years	Sleutel et al. (2006)
Towada, Japan	Rice straw compost	11 t ha <sup>-1</sup>	Potato, maize and soybean in rotation	60 years	Takeda et al. (2005)



Fig. 3 Effects of increasing soil organic matter content and overall soil fertility by soil organic carbon improvement (adapted from Lal, 2006)

organic materials is the calculation both of organic matter supply and plant nutrients. The slow release of these nutrients is responsible for the increase in crop yields in the subsequent years, thus determining the difficulty of quickly evaluating the true agronomic value of these organic materials as amendments. However, there is a considerable variability between experimental techniques, climate, soil type and organic material characteristics, and therefore attention must be paid to generalizing the effects of composts and raw waste application on the soil-plant system.

# 2.1 Effects on Soil Biological, Chemical and Physical Fertility

Van-Camp et al. (2004) found that organic amendments influence soil characteristics by the interdependent modification of biological, chemical and physical properties (Fig. 3). Also, fertility improvement through an effective management of these properties has the capability of optimizing crop production. In this subsection, we selected papers whose primary purpose was to examine the effect of different organic amendments on overall soil fertility. However, it is necessary to take into account that sometimes it was difficult to compare the results, because of the different assessment methods used in the studies.

#### 2.1.1 Biological Fertility

Microbiological and biochemical soil properties are very reactive to small changes occurring in management practices. Therefore, it is possible to use them in a basic analysis for evaluating the effects of the application of different sources and amount of organic matter on soil characteristics during experimental trials. Microorganisms, e.g. bacteria, fungi, actinomycetes and microalgae, play a key role in organic matter decomposition, nutrient cycling and other chemical transformations in soil (Murphy et al., 2007). Since organic carbon (C) is utilized for energy by decomposer microorganisms, its fate is to be either assimilated into their tissues, released as metabolic products, or respired as carbon dioxide  $(CO_2)$ . The macronutrients N, P and sulfur (S), present in the organic chemical structures, are converted into inorganic forms. Subsequently, they are either immobilized and used in the synthesis of new microbial tissues or mineralized

and released into the soil mineral nutrient pool (Baldock and Nelson, 2000). For assimilation by microorganisms of decomposing organic residues, the N has to be assimilated in an amount determined by the C/N ratio of the microbial biomass. More specifically, the amount of N required by the microorganisms is 20 times smaller than that of C. If there are both a low concentration of easily decomposable C compounds and a larger N quantity in respect to that required by the microbial biomass, there will be net N mineralization with release of inorganic N. On the contrary, Corbeels et al. (1999) found that if the amount of N present in the residues is smaller than that required by the microbial biomass, further inorganic N will need to be immobilized from the soil to complete the decomposition process.

It is difficult to distinguish between the direct and the indirect effects of an amendment on the behavior of soil microorganisms. In soils amended with compost or other raw organic materials, even in association with mineral fertilizer N, autochthonous microbiological activity and growth can be stimulated. However, different authors (Ros et al., 2006a; Kaur et al., 2008) suggest that a direct effect from microorganisms introduced with the compost is detectable. Several longlasting experiments have demonstrated that soil biological properties, such as microbial biomass C, basal respiration and some enzymatic activities, are significantly improved by compost treatments. This is particularly evident in the upper layers of the soil because of the added labile fraction of organic matter, which is the most degradable one (Zaman et al., 2004; Ros et al., 2006a, b; Tejada et al., 2006, 2009). Since generally the composts are slowly decomposed in the soil, the continuous release of nutrients can sustain the microbial biomass population for longer periods of time, compared with mineral fertilizers (Murphy et al., 2007). In fact, an interesting residual effect of composts on the microbial activity has often been observed in many experimental seasons after their application, which also results in a longer availability of plant nutrients. Ginting et al. (2003), for example, found 4 years after the last application of compost and manure that the residual effects resulted in 20 to 40% higher soil microbial biomass C compared with the N fertilizer treatment. Research on the effect of different doses of raw municipal solid waste in Mediterranean semiarid conditions, ranging between 65 and 260 t  $ha^{-1}$ ,

demonstrated 17 years after a single application of this organic amendment an average increase of 70% in organic matter content. The water-soluble C fractions also increased by up to 195 tha<sup>-1</sup> application rate, above which they leveled off (Bastida et al., 2008). The authors also found that the enzymatic activities of urease,  $\beta$ -glucosidase, alkaline phosphatase and o-diphenyloxidase associated with humic substances increased significantly in all the amended plots, thus improving soil biochemical quality.

Sixteen years after 1 and 3 t ha<sup>-1</sup> year<sup>-1</sup> sludge application, in a field situated close to Lund in the southern part of Sweden, an increasing trend for substrate induced respiration was found, ranging from 20 to 22%, due to the organic material inputs, and enhanced by from 11 up to 33% acid phosphatase activity involved in the mineralization of organically-bound P (Mats and Lennart, 1999). This behavior follows the increasing rates of sludge addition.

As a general rule, the quantity and quality of organic material added to soils are the major factors in controlling the abundance of different microbial groups and the activity of microorganisms involved in nutrient cycling. Enzyme activity and microbial biomass analysis indicated that microbial properties were stimulated, e.g. microbial biomass C increased to about 100%, more by high rates and composted than by low rates and fresh paper mill residuals (78 t ha<sup>-1</sup> and 22 t ha<sup>-1</sup>, respectively) (Leon et al., 2006).

As regards the effects of the quality of amendment, Monaco et al. (2008) pointed out that microbial activity, measured by the potential soil respiration parameter, gave a reliable and useful indication of the amount of easily decomposable organic C, and this parameter was lower when the C was partially humified before soil input. In fact, after 11 years of repeated applications of different organic materials, soil respiration was in the order of 640.9 < 682.1 <755.1 mg C-CO<sub>2</sub> kg<sup>-1</sup> for farmyard manure, cattle slurry and straw, respectively. The effect of green manure amendments was studied in a 47-year field experiment (Elfstrand et al., 2007). The authors found a higher abundance of bacteria and fungi in the green manure treatment, i.e. 34.3 and 1.8 nmol g soil<sup>-1</sup>, respectively, in respect to the unfertilized one (20.3 and  $0.9 \text{ nmol g soil}^{-1}$ ), measured before maize sowing. Furthermore, a higher fungal/bacterial ratio was noticed, equal to 0.054 compared with farmyard manure

(0.036) and sawdust (0.046) application. This behavior might be attributed to differences in the quality of the organic matter added.

The reviewed results suggest that exogenous organic matter applications to cropland lead to an improvement in soil biological functions, depending on the quantity and type of materials applied.

#### 2.1.2 Chemical Fertility

A considerable number of studies, concerning longterm fertility trials, pointed out that soil organic material applications increased the organic carbon stock and, therefore, increased the cation exchange capacity. This effect was due to the high negative charge of organic matter. This is important for retaining nutrients and making them available to plants (García-Gil, 2004; Ros et al., 2006b; Weber et al., 2007; Kaur et al., 2008). Figure 4 shows the organic carbon increase, as a consequence of several organic amendments' long-lasting application, ranging from 24 to 92%.

Habteselassie et al. (2006a) found that, over a 5year period, the C pool was enhanced by 115% in dairy-waste compost-treated soil. Moreover, the dairywaste compost increased organic carbon by 143 and 54% as compared with ammonium sulfate and liquid dairy-waste treatments, respectively, applied at the same available N level (200 kg Nha<sup>-1</sup>). This C stored in the soil organic matter accounts for approximately 11% of the total amount of C applied. After 3 years of municipal solid waste compost and olive pomace compost application, the total organic carbon significantly increased by 24.0 and 43.2% for cocksfoot and alfalfa plots, respectively, in respect to the unfertilized control, indicating that these amendments positively affected the organic matter (Montemurro et al., 2006). Other authors also showed that municipal-industrial wastes stimulate plant growth, indirectly and with a long-term effect, by improving organic matter (Mantovi et al., 2005; Cherif et al., 2009).

Under 7–36-year fertility experiments in five different rice-based cropping systems, the application of organic amendments at 5–10 t ha<sup>-1</sup>year<sup>-1</sup>, through farmyard manure or compost combined with balanced mineral NPK, increased organic carbon by 10.7% (Mandal et al., 2007).

In a rice–wheat system, farmyard manure application at 20 t ha<sup>-1</sup> showed, after a period of 32 years, higher organic carbon concentration of 17% compared with NPK fertilizers in the 0–15 cm soil layer (Kukal et al., 2009). Nevertheless, the results of repeated applications of either digested sewage sludge over 12 years or farmyard manure for 40 years indicated that such organic amendments were inadequate for restoration of organic matter lost as a consequence of cultivation (Saviozzi et al., 1999). In fact, the amount of organic carbon in the undisturbed site was 120 and 156% higher than that in farmyard and sludge cultivated soils, respectively. On the other hand, contrary to common belief, over a 25-year period of intensive rice-wheat cropping, a depletion of organic



**Fig. 4** Soil organic carbon increases after different longterm organic amendment applications. Note: data taken from: Montemurro et al. (2006) (means of crops); Kaur et al. (2008) (at 0–15 cm depth); Kukal et al. (2009) (means of crops; at 0–15 cm

depth); Mandal et al. (2007) (means of 5 trials' data); Meng et al. (2005); Monaco et al. (2008); Triberti et al. (2008). SOC = soil organic carbon; MSW = municipal solid waste; FYM = farm-yard manure

carbon did not occur, but rather an improved organic carbon concentration of 38% (Benbi and Brar, 2009).

In terms of sustainability, only farmyard manure fertilization maintained the total organic carbon level of 40 t C ha<sup>-1</sup>, measured in the top soil layers at the start of a 40-year experiment, while the average total organic C depletion was 23% with liquid manure and mixed fertilization treatments, 43% with mineral fertilizers alone and 51% in the control (Nardi et al., 2004). Furthermore, the presence of weakly acidic chemical functional groups on organic molecules makes organic matter an effective buffer, as supported by the findings of García-Gil (2004). These authors observed a long-(9 years) and short-term improvement in the soil humic acid buffering capacity in municipal solid waste compost-amended soils, derived from a residual effect of a single application and cumulative effects from repeated ones. These distributions will favor the general soil fertility status and crop production.

There are reports in the literature of long-term compost and manure application both increasing (Eghball, 2002; García-Gil et al., 2004; Butler and Muir, 2006) and decreasing (Meng et al., 2005; Bastida et al., 2008; Bi et al., 2008) the pH of soils, depending on their initial pH and organic residues. Butler and Muir (2006) observed that soil pH increased on average by 0.5 units as the dairy manure compost rate doubled in magnitude from 11.2 to 179.2 t ha<sup>-1</sup>.

As previously explained, with long-term compost use the residual effects on crop production and soil properties can last for several years, since only a fraction of the N and other nutrients becomes available for plants in the first year after spreading (Hartl et al., 2003; Eghball et al., 2004). As an estimation of the the available N from compost treatment in the first year of application, Tittarelli et al. (2007) mentioned a release of only 30–35% of the total N content. The N release from compost will mostly occur in the first two years after application, suggesting that a distribution frequency of once in every second year may be better than other application strategies, especially with higher rates (Zhang et al., 2006).

More specifically, it is well known that many microorganisms convert organic N into inorganic N forms by mineralization. A large number of authors confirmed that N mineralization from compost is very limited in the short term. However, there is a significant residual effect from the cumulative applications which becomes visible later after 4–5 years, resulting in deferred higher N availability and yields (Eghball, 2002; Blackshaw et al., 2005; Barbarick and Ippolito, 2007; Leroy et al., 2007). Regular addition of organic material to soil for more than 10 years, through compost or manures, enhanced both soil C and N stocks and resulted in build-up of N, indicating a physical protection of this nutrient within macroaggregates (Whalen and Chang, 2002; Meng et al., 2005; Mallory and Griffin, 2007; Sodhi et al., 2009). According to Hartl and Erhart (2005), the organic N content increases by about 10% compared with the control, in the upper 30 cm of soil, after 10 years of compost treatments. This result was complemented by significant increases in organic C, of 22%, indicating that the organic N was tied up in organic matter. After 4 years of vegetable compost applications, significantly higher soil total N concentration was observed on compost plots compared with plots without it (Nevens and Reheul, 2003).

The C/N ratio of organic material can be used as a good indicator of nutrient supply. Tejada et al. (2009) showed an optimum balanced C/N ratio (10-12) for soils amended with composts originating from leguminous residues, due to organic matter mineralization overcoming immobilization. These findings were not in accordance with those of Weber et al. (2007), who found the C/N ratio clearly increased, from 10.7 up to 22.2, in the third year after municipal solid waste compost application. This behavior can be explained by a depletion of N reserve, probably because of plant N uptake. However, there are often other explanations for such an increase in the C/N ratio. In fact, it is well known that when a compost that has a high C/N ratio is added to soil, the microbial population competes with plants for soil N, thus immobilizing it (Amlinger et al., 2003).

Soil available potassium (K) content increased on average by 26%, as compared with control, in 5-year compost treatments derived from organic household wastes and yard trimmings (Hartl et al., 2003). These treatments are a rich source of K, probably due to the large proportion of woody plant material and kitchen refuse in the raw material.

As regards the P from organic amendments, He et al. (2001) reported that compost applications can increase plant-available P in the soil. The biosolidsmunicipal solid waste co-compost, applied once in 4 years, has also been found to effectively supply P to soil at 0–15 cm depth. The soil extractable P concentration increased on average from 7.2 to 86 mg kg<sup>-1</sup> soil with enhanced application rates from 0 to 200 t ha<sup>-1</sup> (Zhang et al., 2006). Furthermore, Eghball (2002) suggested that 4-year beef cattle manure and composted manure application based on N needs of corn could eventually result in soil accumulation of P, since the manure or compost N/P ratio is usually smaller than the corn N/P uptake ratio.

Our overall literature analysis demonstrates that several organic amendments' long-lasting applications enhanced soil available potassium, extractable phosphorous and organic carbon content, and resulted in deferred N availability.

#### 2.1.3 Physical Fertility

Aggregate stability is a keystone factor in questions of soil physical fertility and can be enhanced by means of an appropriate management of organic amendments, which can maintain an appropriate soil structure. This agronomic procedure could improve pore space suitable for gas exchange, water retention, root growth and microbial activity (Van-Camp et al., 2004). Soils rich in organic matter are less prone to erosion processes than soils with low content, such as those which predominate in arid and semi-arid areas (Durán Zuazo and Rodríguez Pleguezuelo, 2008).

The topic of soil structural stability has been reviewed recently by Abiven et al. (2008). Their literature analysis validated the conceptual model proposed by Monnier (1965). This author considers different patterns of temporal effects on aggregate stability, depending on the nature of the organic inputs. Easily decomposable products have an intense and transient effect on aggregate stability, while more recalcitrant ones, such as lignin and cellulose, have a lower but longer-lasting effect.

Results of Albiach et al. (2001) pointed out that organic matter and carbohydrates appeared to be the parameters most closely related to structural stability of soil aggregates obtained with five applications of a municipal solid waste compost. Another 5-year field trial confirmed these findings, suggesting that municipal solid waste compost, applied every 2 years, can be used to increase soil aggregate stability by 29.3% in respect to the control, thus improving soil resistance to water erosion (Annabi et al., 2006).

The organic matter stabilizes soil structure by at least two different mechanisms: by increasing the inter-particle cohesion within aggregates and by enhancing their hydrophobicity, thus decreasing their breakdown, e.g. by slaking. More specifically, the increase in soil microbial activity, especially due to the addition of composted residues, could be responsible for the increase in soil structural stability (Van-Camp et al., 2004). The relationships between soil biological activity and the functioning of soil are very complex. According to Abiven et al. (2008), several biological binding agents have been recognized as accountable for aggregation and aggregate stability. Polysaccharides synthesized by microorganisms, particularly at the beginning of the organic matter decomposition, tend to adsorb the mineral particles and increase their inter-cohesion. Conversely, products rich in humic compounds, such as manures or composts, would also be expected to increase aggregate hydrophobicity of clays. This has mainly been proved in long-term trials. In fact, after 16 years of either farmyard manure or crop waste applications, substantial improvements in soil physical properties have been noticed, markedly aggregate stability and water retention, due to the increased concentration of humic colloids in soil (Dorado et al., 2003). In particular, the authors found that the structural instability index decreased by 2.5 units with respect to control plots.

Tejada et al. (2009) observed that three different composts, consisting of leguminous plants, nonleguminous plants and the combination of both plant residues, applied at rates of 7.2 and 14.4 t organic matter ha<sup>-1</sup> during a period of 4 years, had a positive effect on soil physical properties. More specifically, at the end of the experimental period and at the highest rate, the soil structural stability was the highest in the non-leguminous plant compost treatment (28.3%), followed by the combined one (22.4%). This result was due to greater amounts of humic acids provided to the soil, 63.6 and 59.5 g kg<sup>-1</sup>, respectively, which are then directly involved in clay–organic complex formation.

Another recent study showed that, after 10 cycles of rice–wheat cropping, the amount of water-stable aggregates was significantly higher in plots amended with rice straw compost at 8 t ha<sup>-1</sup> to both rice and wheat, as compared with inorganic fertilizers (Sodhi et al., 2009). The authors also suggested that this higher amount of water-stable aggregates can be ascribed to the regular addition of organic matter to soil, resulting in enhanced microbial activity and production of microbial decomposition products, which helps

with binding of aggregates. Leon et al. (2006) found that the application of medium and high rates, equal to 38.1 and 78.4 t ha<sup>-1</sup>, of composted paper mill residuals over 4 years, caused an increase in amounts of water-stable aggregates on average of 25% compared with the control. There was no significant difference ( $P \le 0.05$ ) between the two compost rates tested, suggesting that the lowest amount of amendment was enough to maximize water-stable aggregation.

These positive results of long-term compost application agreed with the findings reported by Tejada et al. (2008). The authors observed that composted leguminous plants, alone or mixed with beet vinasse, an agroindustrial by-product, had a positive impact on soil structural stability, which increased by 5.9 and 10.5% compared with the unfertilized treatment, respectively. Conversely, fresh beet vinasse application decreased soil structural stability by 16.5% in respect to the control, probably because high quantities of destabilizing monovalent cations were introduced into the soil by this organic material.

Whalen and Chang (2002) noted that 25 years of annual beef cattle manure applications, at rates of more than 30 t  $ha^{-1}$  year<sup>-1</sup> to dry land soils and at more than 60 t  $ha^{-1}$  year<sup>-1</sup> to irrigated soils, can shift the aggregate size distribution from larger (>12.1 mm) to smaller (< 2.0 mm) dry-sieved aggregates, because of dispersive agents in the manure. Consequently, many years of continuous manure applications could raise the risk of wind erosion because of a greater proportion of the soil small aggregates. Moreover, different authors have shown typical decreases in soil bulk density, on average of 15%, after long-term compost, farmyard manure or digested sewage sludge agricultural use, suggesting that the addition of composted and fresh organic matter facilitated, as a consequence, the soil porosity connected to soil bulk density (Saviozzi et al. 1999; Meng et al., 2005; Tejada et al., 2008; 2009). Porosity is a measure of the size and system of voids in the soil matrix, affecting both aeration and water movement. On the other hand, Eghball (2002) reported that soil bulk density was unaffected by 4-year manure or compost application.

Leon et al. (2006) found a strong correlation (r = 0.65,  $P \leq 0.001$ ) of total C with soil moisture content after application of paper mill residual by-products, corroborating that high C level in soils increases the water-holding capacity because of the effect of organic matter on soil aggregation. This increase pro-

vides more available water to plants, also helping with resistance to drought.

On the basis of the information presented in this subsection, it can be summarized that repeated applications of organic amendments can increase soil physical fertility, mainly by improving aggregate stability. In general, a rise in the above investigated overall soil fertility influences crop yield response, as framed in the following subsection. Selected studies are summarized in Table 2, focusing on the effects of organic amendments on yields.

# 2.2 Effects of Organic Amendments on Plant Nutrition and Yielding Responses

The N dynamics in compost-amended soils could be affected by different site-specific factors, e.g. compost matrices, composting conditions, climate, soil properties and management practices. It can be generally assumed that the prompt availability of N to plants is low, as already explained in this review, since the majority of total compost N is bound to the organic N-pool (Amlinger et al., 2003). In fact, the greatest total N content in compost is not readily available, but it can be mineralized and subsequently taken up by plants or immobilized, denitrified and/or leached.

It is important to take into account that the slow release of nutrients from compost or green manures should be adequately controlled to match temporal crop demand with nutrient supply. The increase in the N-use efficiency decreases loss to leaching when there is wide drainage, e.g. during the fall-winter period, or volatilization (Tilman et al., 2002). The plant-available N in the application year was expected to be higher for fresh manure than for stable composted manure. In fact, in a 4-year experiment, it ranged from 26 to 67 and from 12 to 18 kg ha<sup>-1</sup> for fresh and stabilized manure, respectively (Blackshaw et al., 2005). Furthermore, Zhang et al. (2006) reported that the amount of N used by crops from municipal solid waste compost for barley, wheat and canola, at two different sites, was 11, 3, 1 and 2% for the first and subsequent 3 years. These results indicate the complexity of estimating N release from different composts and its relationship with plant N uptake. Barbarick and Ippolito (2007) showed that

Crop andAmendments and application ratetrial periodrateWheatMSW composts C1 (40 t ha <sup>-1</sup> ); (5 years)(5 years)and C2 (80 t ha <sup>-1</sup> ); (40 t ha <sup>-1</sup> ); MSW + Cf; FYM - (40 t ha <sup>-1</sup> ); MSW + Cf; FYM - (10 wastecomposted (two rates: $5$ and 10 t ha <sup>-1</sup> year <sup>-1</sup> )Wheat-Municipal-industrial wastewater sugarbeetMinize-S and 10 t ha <sup>-1</sup> year <sup>-1</sup> )Naize(10 years)(10 years)(16 t ha <sup>-1</sup> year <sup>-1</sup> )potatoes(9 t ha <sup>-1</sup> year <sup>-1</sup> )(7 years)(10 years)(7 years)and C2 (45 t ha <sup>-1</sup> ); (7 years)Tall fescueThree food waste compost with: (7 years)(7) years)(1) yard trimmings, + mixed paper waste; 3) Wood waste + sawdust (each at 155 t ha <sup>-1</sup> )Silage(1) Low dairy-waste compost vith: (7 years)Silage(1) Low dairy-waste compost vith: (2) high dairy-waste compost vith: (2) high dairy-waste compost vith: (2) high dairy-waste compost				
Crop andAmendments and application rateWheatMSW composts C1 (40 t ha <sup>-1</sup> ); FYM (5 years)WheatMSW composts C1 (40 t ha <sup>-1</sup> ) and C2 (80 t ha <sup>-1</sup> ); FYM (40 t ha <sup>-1</sup> ); MSW + Cf; FYM - Mudge:Wheat-Municipal-industrial wastewater sludge:wheat-Municipal-industrial wastewater sludge:wigarbeet(1) anaerobically digested, rotation(12 years)(11) composted (two rates: $5$ and 10 t ha <sup>-1</sup> year <sup>-1</sup> )Cereals andBiowaste compost: C1 $0$ t ha <sup>-1</sup> year <sup>-1</sup> )Detatoes(16 t ha <sup>-1</sup> year <sup>-1</sup> )(10 years)(16 t ha <sup>-1</sup> year <sup>-1</sup> )(7 years)(10 years)(7 years)(10 years)(7 years)(1) yard trimmings: and C2 (45 t ha <sup>-1</sup> ); cattle slurryTall fescueThree food waste compost with: (7 years)(1) yard trimmings: and C2 (45 t ha <sup>-1</sup> ); cattle slurrySilage(1) yard trimmings: awdust (each at 155 t ha <sup>-1</sup> ) of years)Silage(1) Low dairy-waste compost available N);Silage(1) Low dairy-waste compost available N);(5 years)(2) high dairy-waste compost trate (200 kg N ha <sup>-1</sup> );			increase (*)	
trial periodrateWheatMSW composts C1 (40 t ha <sup>-1</sup> ); FYM(5 years)and C2 (80 t ha <sup>-1</sup> ); FYM(5 years)and C2 (80 t ha <sup>-1</sup> ); FYMWheatMunicipal-industrial wastewaterwheatMunicipal-industrial wastewaterwheatsugarbeet(12 years)(1) anaerobically digested,rotation(1) dewatered and(12 years)(11) composted (two rates:5 and 10 t ha <sup>-1</sup> year <sup>-1</sup> )Cereals andBiowaste compost: C1potatoes(16 t ha <sup>-1</sup> year <sup>-1</sup> ) and C3(10 years)(16 t ha <sup>-1</sup> year <sup>-1</sup> ) and C3(7 years)(10 years)(7 years)(1) yard trimmings;(7 years)(1) yard trimmings;(7 years)(1) yard trimmings;Silage(1) Low dairy-waste compost with:(5 years)(2) high dairy-waste compostsudust (each at 155 t ha <sup>-1</sup> );(5 years)(2) high dairy-waste compost(5 years)(2) high dairy-waste compost	ion Sub-treatment or	Highest yield	(% $t^{-1}$ of total	
WheatMSW composts C1 (40 t ha <sup>-1</sup> ); FYM (5 years)MSW composts C1 (40 t ha <sup>-1</sup> ); mSW + Cf; FYM + (40 t ha <sup>-1</sup> ); MSW + Cf; FYM + wheat-Wheat-Municipal-industrial wastewater sludge:Wheat-Municipal-industrial wastewater sludge:Wheat-Municipal-industrial wastewater sludge:Wheat-Municipal-industrial wastewaterWheat-Municipal-industrial wastewaterWheat-Municipal-industrial wastewaterWidge:S and 10 t ha <sup>-1</sup> year <sup>-1</sup> )Cereals andBiowaste composted (two rates: 5 and 10 t ha <sup>-1</sup> year <sup>-1</sup> )Deptatoes(10 years)(10 years)(16 t ha <sup>-1</sup> year <sup>-1</sup> ), c2 (10 years)MaizeVFG compost: C1 (23 t ha <sup>-1</sup> year <sup>-1</sup> )MaizeVFG compost: C1 (23 t ha <sup>-1</sup> )MaizeVFG compost: C1 (2) yard trimmings; and C2 (45 t ha <sup>-1</sup> ); cattle slurryTall fescue(1) yard trimmings; paper waste; 3) Wood waste + sawdust (each at 155 t ha <sup>-1</sup> ) of (5 years)Silage(1) Low dairy-waste compost available N);Silage(1) Low dairy-waste compost available N);(2) high dairy-waste compost rate (20 0 kg N ha <sup>-1</sup> );	comment	$(t ha^{-1})$	C applied)	Reference
Wheat- maize-Municipal-industrial wastewater maize- sugarbeet (1) anaerobically digested, rotation (12) years) (12) years)Municipal-industrial wastewater sludge: 5 and 10 tha <sup>-1</sup> year <sup>-1</sup> ) (23 tha <sup>-1</sup> year <sup>-1</sup> ), C2 (10 years) (10 years)Municipal-industrial wastewater s and 10 tha <sup>-1</sup> year <sup>-1</sup> ) (23 tha <sup>-1</sup> year <sup>-1</sup> ) and C3 (23 tha <sup>-1</sup> year <sup>-1</sup> ) plus 5 treatments with combined fertilization maizeMaize (7 years)VFG compost: C1 (22.5 tha <sup>-1</sup> ) and C2 (45 tha <sup>-1</sup> ); cattle slurryTall fescue (7 years)Three food waste compost with: cattle slurrySilage (5 years)(1) Low dairy-waste compost available N); (2) high dairy-waste compost available N);	na <sup>-1</sup> ) Higher HM M content in MSW Sf; FYM + Cf compost treatments	C2 <sup>(**)</sup> : 60.2 t ha <sup>-1</sup> ; C2Cf: 61.9 t ha <sup>-1</sup>	17	Cherif et al. (2009)
Cereals andBiowaste compost: C1potatoes $(9 t ha^{-1} year^{-1}), C2$ $(10 years)$ $(16 t ha^{-1} year^{-1})$ and C3 $(10 years)$ $(16 t ha^{-1} year^{-1})$ and C3 $(23 t ha^{-1} year^{-1})$ $plus 5 treatments withmaizeVFG compost: C1 (22.5 t ha^{-1})(7 years)and C2 (45 t ha^{-1});cattle slurryThree food waste compost with:(7 years)(1) yard trimmings; + mixedpaper waste; 3) Wood waste +sawdust (each at 155 t ha^{-1} rate)Silage(1) Low dairy-waste compost(5 years)(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(5 years)(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost(2) high dairy-waste compost$	water Mineral fertilizer dressings: four ed, increasing rates of urea, plus s: superphosphate 1)	I (at 10 t ha <sup>-1</sup> year <sup>-1</sup> ) for sugarbeet: $60.5 \text{ t ha}^{-1}$ II (at 5 t ha <sup>-1</sup> year <sup>-1</sup> ) for wheat: $6.23 \text{ t ha}^{-1}$ ;	10; 11	Mantovi et al. (2005)
Maize     VFG compost: C1 (22.5 t ha <sup>-1</sup> ); and C2 (45 t ha <sup>-1</sup> ); cattle slurry       Tall fescue     Three food waste compost with: (7 years)       (1) yard trimmings;       (1) yard trimmings;       (2) yard trimmings;       (2) yard trimmings;       (1) Low dairy-waste compost with: (2) yard trimmings;       (1) yard trimmings;       (1) vard trimmings;       (1) vard trimmings;       (1) vard trimmings;       (1) vard trimmings;       (2) yard trimmings;       (1) vard trimmings;       (1) Low dairy-waste compost comfield       (2) vard dairy-waste compost (5 years)       (2) high dairy-waste compost trate (200 kg N ha <sup>-1</sup> );	Mineral fertilizer: three increasing rates (N1; N2; N3)	N3 + C1: 5.2 t ha <sup>-1(**)</sup> ; N2 + C1: 4.9 t ha <sup>-1</sup>	15	Erhart (2005)
Tall fescueThree food waste compost with:(7 years)(1) yard trimmings;(7 years)(2) yard trimmings;(2) yard trimmings + mixedpaper waste; 3) Wood waste +sawdust (each at 155 t ha <sup>-1</sup> rate)sawdust (each at 155 t ha <sup>-1</sup> rate)Silage(1) Low dairy-waste compostcornfieldrate (to provide 100 kg ha <sup>-1</sup> of(5 years)available N);(2) high dairy-waste compostrate (200 kg N ha <sup>-1</sup> );	ha <sup>-1</sup> ) Mineral fertilizer	C2 plus cattle slurry: 22.4 t ha <sup>-1</sup>	5 (from compost C applied)	Leroy (2007)
<ul> <li>Silage (1) Low dairy-waste compost cornfield rate (to provide 100 kg ha<sup>-1</sup> of (5 years) available N);</li> <li>(2) high dairy-waste compost rate (200 kg N ha<sup>-1</sup>);</li> </ul>	with: Two composting methods: (I) aerated ked static pile; waste + (II) aerated turned ha <sup>-1</sup> rate) windrow	(1) $79 \text{ t ha}^{-1}$	0.2(***)	Sullivan et al. (2003)
<ul> <li>(3) low liquid dairy-waste rate</li> <li>(100 kg N ha<sup>-1</sup>);</li> <li>(4) high liquid dairy-waste rate</li> <li>(200 kg N ha<sup>-1</sup>)</li> </ul>	oost Other treatments: ha <sup>-1</sup> of ammonium sulfate at 100 kg N ha <sup>-1</sup> post and 200 kg N ha <sup>-1</sup> : rate : rate e rate	(1) 24.6 t ha <sup>-1</sup>	9.7 (in the last year of the experiment)	Habteselassie et al. (2006a)

in vields (selected data) 5 1040 mbu 5 2 J. ary of the effects six applications once every second year of anaerobically digested biosolids, in a 2-year wheat–fallow rotation, provided about 9 kg N t<sup>-1</sup> biosolids. Sullivan et al. (2003) noted that 7 years of one-time food waste compost application consistently increased tall fescue N uptake by a total of 294 to 527 kg ha<sup>-1</sup>. Therefore, the increase in grass uptake due to amendment was 15 to 20% of compost N applied. However, compost N immobilization/mobilization is predominantly linked to the degradability and balance of soil C pools, as will be better explained in Subsection 3.3 of this review.

Moreover, Eghball et al. (2004) found that the lowest rate of P (125 kg ha<sup>-1</sup>) was applied for the annual P-based beef cattle manure whilst the greatest amount (594 kg P ha<sup>-1</sup>) was applied for the biennial N-based cattle manure compost treatment. The latter resulted in P leaching to a soil depth of 45 to 60 cm. On the contrary, the P-based application was environmentally sound, since it provided nutrients for the crop, while maintaining the amended soil P at a level similar to the untreated control.

#### 2.2.1 Yield Response

There are three possible scenarios relating crop yield or agronomic productivity to organic C content of soil: (i) increase in crop yield as a consequence of organic carbon pool enhancement; (ii) no or little decrease in crop yield with reduction in the organic carbon pool, and (iii) increase in crop yield with decrease in the organic carbon pool (Lal, 2006). These apparently conflicting responses depend on several factors such as the previous organic carbon pool, soil management, and use of chemical fertilizers and organic amendments.

As demonstrated by several long-term experiments on crop nutrition and yielding responses, the benefits of increased organic matter content will differ on the basis of the rate supplied. In a 5-year trial, Hartl et al. (2003) found that every second year spreading of 40 t ha<sup>-1</sup> biowaste compost, from source-separated organic household waste and yard trimmings, resulted in slightly higher (9%) rye yields than other rates. This result suggested that beneficial use depends on choosing the best amount and frequency of compost application.

Frequently, the best rate is the highest, as supported by Butler and Muir (2006), who observed the greatest tall wheatgrass dry matter yield with the highest composted dairy manure rate of about 180 t ha<sup>-1</sup>. More specifically, forage yield increased from 32 to 96% with 11.2 and 179.2 t compost ha<sup>-1</sup>, respectively, in the first of two growing seasons. Sullivan et al. (2003) highlighted the long-lasting effect of a one-time high rate (155 t ha<sup>-1</sup>) food waste compost application in providing slow-release N for crop growth, over a 7-year period. These results agreed with the findings of Habteselassie et al. (2006a), who found that soils with about 100 t ha<sup>-1</sup> dairy-waste compost maintained N supply to the plants through continuous mineralization, as shown by available inorganic N pools, silage corn yield and plant N content analysis.

Wheat grain yield was enhanced on average by 246%, in respect to the control, with a high rate  $(80 \text{ t ha}^{-1})$  of municipal solid waste compost applied annually over 5 years, alone or combined with mineral fertilizer, reaching 60.2 and 61.9 t  $ha^{-1}$ , respectively (Cherif et al., 2009). Similarly, Erhart et al. (2005) investigated the agronomic performance of biowaste compost, on cereals and potatoes, showing that the highest application rate equal to 23 t  $ha^{-1}year^{-1}$ increased crop yields by 10% compared with the unfertilized control, on the average of 10 years. These findings are in contrast with those of Mantovi et al. (2005). In repeated sewage sludge applications for 12 years on a winter wheat-maize-sugarbeet rotation, the composted sludge gave a significantly lower yield, equal to 51.8 t ha<sup>-1</sup>, for sugarbeet with higher dose spreading compared with non-composted biosolid treatment yield (60 t ha<sup>-1</sup>, on average). Moreover, residual effects of amendments on both soil properties and crop production can last for several years, as referred to in the preceding subsection of this review. For instance, in a 7-year trial Eghball et al. (2004) confirmed that available P concentration in the soil surface can contribute to corn crop P uptake for more than 4 years after the last biennial N-based compost application, being 241% higher than the control.

The examined literature pointed out that the use of composts or of other organic amendments in combination with mineral fertilizers enhanced crop yield in many cropping systems over more than 10 years, compared with compost and amendments alone (Ros et al., 2006a; Bi et al., 2008). The subject has recently been discussed by Montemurro (2009), who found that the multiple application of municipal solid waste compost associated with mineral fertilizer increased wheat yield by 8% compared with mineral N alone and induced a more productive stability and N uptake throughout the years. Furthermore, maize grain yield was the highest where farmyard manure at  $10 \text{ t ha}^{-1}$  was applied along with recommended NPK fertilizer for 34 years, under a maize-wheat cropping system (Kaur et al., 2008). In a study by Sleutel et al. (2006), 41 years of application of 35 t farmyard manure  $ha^{-1}$ , plus an equivalent amount of NPK in mineral fertilizer, every 5 years in a wheat-wheat-sugarbeet-maize-maize crop rotation, increased grain yield by 124 and 55% for maize and wheat in respect to the control. Moreover, in a 7-year study, Leroy et al. (2007) investigated vegetable-fruitgarden-waste compost combined with cattle slurry applied at both 22.5 t  $ha^{-1}$  yearly and 45 t  $ha^{-1}$  every other year. Both the application strategies resulted in 25 to 43% higher maize yields in respect to the two organic amendments provided alone. On the contrary, Edmeades (2003), after reviewing 14 long-term soil fertility trials, concluded that, despite the positive effects of organic manures on soil biological and physical properties, no advantage to crop yields with respect to the application of the same amounts of nutrients as mineral fertilizers was found. However, it is necessary to note that in organic farming systems the use of mineral fertilizers is not a possible alternative. After all, the analyzed soil management practices used in sustainable farming systems have potential for producing comparable yields to conventional farming ones. Poudel et al. (2002) reported that the average tomato fruit and corn grain yields, for a 5-year trial period, were 71.0 and 11.6 t ha<sup>-1</sup>, respectively, both not significantly different among organic, low-input and conventional farming systems.

From all the above discussion, there is clear evidence that the best agronomic performance of compost, particularly if combined with mineral fertilizers, is often obtained with both high rates and frequency of applications, leading to residual effects as a slowrelease nitrogen fertilizer.

#### 2.2.2 Quality Response

From the yield quality viewpoint, the reviewed longterm research showed, for example, that sourceseparated organic waste compost as well as mixtures of sugarbeet vinasse composted with other agroindustrial solid wastes did not adversely affect the quality of products (Madejón et al., 2001; Erhart et al., 2005). In particular, winter rye protein concentration was similar in compost and mineral fertilized treatments, and the nitrate concentration of potato tubers in compost treatments was not significantly higher than in the unfertilized control (Erhart et al., 2005). The crop quality in some cases was even improved by compost fertilization. In one above-mentioned study, the partial substitution of mineral with organic N not only did not reduce the quality of wheat, with respect to mineral fertilizer, but also increased the protein content by 6% in comparison with the unfertilized control (Montemurro, 2009). In another study the wheat grain quality, expressed by its apparent specific weight as means of 12 cropping years, was worsened by excessive N supply, so high rates  $(10 \text{ t ha}^{-1} \text{ year}^{-1})$  of liquid and dewatered sludge were particularly detrimental, while compost was safer (Mantovi et al., 2005). In fact, a downward trend was observed as 74.7 > 73.5 >72.8 kg  $hL^{-1}$  for composted, dewatered and liquid sludge, respectively.

Therefore, considering that stabilized organic amendment application does not reduce the crop yield quality, as reviewed here, but can even enhance it, their use can appear more profitable.

# 3 Environmental Impact and Sustainability of Organic Amendments

As recently highlighted by Lichtfouse et al. (2009), while conventional agriculture is driven almost solely by productivity and profit, sustainable agricultural systems aim at developing new farming practices that are also safe and do not degrade the environment. By taking this into account, and from the previous discussions, it is possible to suggest that appropriate organic matter management is a fully sustainable pattern, because it fulfils three requisites: sustainability of resources, human health and economic sustainability. Conversely, organic amendments that can improve organic matter and the linked soil fertility can also be a source of environmental pollution, especially when they are improperly used. In fact, when the application rates of manure are calculated on the N crop requirement, the amount of P added often exceeds the plant P requirement, resulting in soil P accumulation (Miller and Miller, 2000). According to Komatsuzaki

and Ohta (2007), soil management strategies for sustainable agriculture should focus not only on increasing organic matter in the soil, but also on the uptake or stocking of soil residual nutrients in such a way to prevent excess nutrient leaching into the groundwater.

The cropland application of immature compost can produce environmental and agronomic problems. In fact, if the organic material has not been sufficiently stabilized, its application increases ammonia volatilization, decreases the soil oxygen concentration, produces some phytotoxic compounds and immobilizes soil mineral N. Therefore, organic products of high quality must be produced and their stability must be accurately assessed.

With regard to toxic elements, e.g. heavy metals, their accumulation in soils is the most often cited potential risk, particularly for biosolid waste compost use. On the other hand, there is increasing positive evidence of the impact that composts and wastes can have on soil C sequestration, as we will discuss in the following pages.

# 3.1 Organic Matter Evolution and Soil Carbon Sequestration

In recent years, global concern over increased atmospheric  $CO_2$  and methane (CH<sub>4</sub>) emissions has raised interest about the potential role of soils as a source or sink of C and in studying organic matter dynamics and related C sequestration capacity. In fact, organic matter varies in both decomposition rate and turnover time. and the soil C pool can be a source or sink for the atmospheric pool, depending on land use and management (Van-Camp et al., 2004). When organic materials, such as composted wastes, are added to soil, at least a share of their organic C is decomposed producing  $CO_2$ , which is a good indicator of the decomposition rate, while another part is sequestered in the soil.

For the purpose of this review, the term "carbon sequestration" is used according to Lal (2004, 2007), implying the transfer of atmospheric CO<sub>2</sub> into the soil C pool through: (i) humification of crop residues and other wastes added to the soil, (ii) formation, in arid and semi-arid regions, of secondary carbonates or leaching of bicarbonates into the groundwater, so CO<sub>2</sub> thus adsorbed is not immediately re-emitted, (iii) formation of organo-mineral complexes which encapsulate C and protect it against microbial activities, and (iv) translocation of organic carbon into the sub-soil, that can move it away from the zone of disturbance by plowing and other agronomic practices, minimizing the risks of being removed by erosional processes.

Article 3.4 of the Kyoto Protocol emphasizes the agricultural role in  $CO_2$  sequestration in tilled soils and advocates sustainable cropping techniques for this purpose. These practices primarily include the reduction of soil disturbance and the optimization of water-use efficiency. However, soil incorporation of organic materials and N fertilization can also influence C dynamics (Triberti et al., 2008). Any attempt to enrich the organic carbon reservoir through sequestration of atmospheric C will help to manage global warming.

According to Feller and Bernoux (2008) findings, it is essential to consider that when an organic waste contains a high percentage of fossil carbon, the C in soils originating from this fraction should not be referred to as sequestered C, i.e. originating from the atmosphere, but as "stored C".

As explained by Mondini and Sequi (2008), it is also important to consider that the reduction in the soil of the C pool contributes both to an atmospheric enrichment in  $CO_2$  concentration and an involution in soil fertility. This reduction also induced the onset of degradative processes such as erosion, salinization, desertification, compaction, nutrient deficiency, etc. However, whereas the exact magnitude of the historic loss of organic carbon may be debatable, Lal (2004) suggested that the troubling process of organic carbon depletion can be reversed.

The effects of organic amendments on organic carbon increase should be studied through long-term field experiments, as already seen in Subsection 2.1.2. of this review, because of the long time needed to attain a new equilibrium after environmental change in the organic matter.

Triberti et al. (2008) reported that 29 years after the start of a trial comparing cattle manure, cattle slurry and crop residues with mineral fertilization, the cattle manure gave the quickest organic carbon stock build-up, 0.26 t organic carbon ha<sup>-1</sup> year<sup>-1</sup>. Each t dry matter ha<sup>-1</sup> of applied cattle manure, containing 33.1% of organic C, increased the organic carbon stock by 27 kg C ha<sup>-1</sup> in the 0–0.4 m soil layer. This increase corresponded to the highest sequestration efficiency, equal to 8.1% of the C added, due to its low degradability, as compared with 3.8 and 3.7% in the instances of cattle slurry and cereal crop residues. On the whole, the recycling of cattle farming by-products for  $CO_2$  sequestration purposes poses environmental risks, such as pollution by nitrate N (NO<sub>3</sub>–N), that can limit its sustainability in fertile cropland of developed countries.

About 25 and 36% of applied manure and compost C remained in the soil after 4 years of application, indicating greater C sequestration with composted than non-composted manure (Eghball, 2002). According to Sodhi et al. (2009), a 10-year application of rice straw compost, either alone or in combination with inorganic fertilizers, results in C sequestration in macroaggregates. In fact, with the application of 8 t compost ha<sup>-1</sup>, the C concentration in the 1–2 mm size fraction increased by 180 to 191%, respectively, over unfertilized control. From trials still in progress for more than 10 years, there is evidence that the organic carbon sequestration rate increased more due to farmyard manure and composted farmyard manure in comparison with mineral fertilizers or other organic materials (Monaco et al., 2008; Kukal et al., 2009).

Mandal et al. (2007) observed that the total quantity of soil C sequestered over a long period was linearly related to the cumulative crop residue C inputs, and the rate of the conversion to organic carbon was higher in the presence of added organic materials, i.e. 6.9% of each additional t C input ha<sup>-1</sup>, than in their absence (4.2%). Indeed, it has been calculated that the rate of organic carbon sequestration is on average 0.3 to 0.5 t C ha<sup>-1</sup> year<sup>-1</sup> under intensive agricultural practices (Lal, 2007). As reported by Mondini and Sequi (2008) organic matter is the largest C stock of the continental biosphere, with 1550 billion tons. On the time scale of several decades, in arable soils which receive high organic materials, relatively less organic matter is stabilized either by association with the silt plus clay mineral fractions, or by its inherent biochemical recalcitrance (Sleutel et al., 2006).

Shindo et al. (2006) reported that continuous compost application in a field subjected to 25 years of double cropping could increase both the amounts of fulvic and humic acids, and the total humus content. From the organic carbon sequestered quality standpoint, Nardi et al. (2004) found that, over 40 years, farmyard manure fertilization improved by 116% the production of humus with a high degree of polycondensation, a high-quality fraction usually linked to soil fertility. Conversely, the absence of organic fertilizer inputs determined the opposite, with a higher percentage of non-complex and lightweight humus.

Besides, the soil C sequestration should not be restricted to a mere quantification of C storage or  $CO_2$ balance. All greenhouse gas fluxes must be computed at the plot level in C–CO<sub>2</sub> or CO<sub>2</sub> equivalents, incorporating as many emission sources and sinks as possible across the entire soil-plant system (Feller and Bernoux, 2008). For example, results of Ginting et al. (2003) showed that fluxes of CH<sub>4</sub> and nitrous oxide (N<sub>2</sub>O) were nearly zero after 4 years of manure and compost applications. This evidence indicates that the residual effects had no negative influence either on soil C and N storage, or global warming. However, there is a lack of information and a strong need to further understand the greenhouse gas fluxes as related to organic material input dynamics in the soil.

The reviewed results suggest that the process of organic carbon depletion can be reversed by long-term organic amendment application. Soil carbon sequestration should be considered as a "win–win" strategy for increasing soil fertility and preventing soil erosion processes (Sánchez-Monedero et al., 2008).

# 3.2 Heavy Metal Concentration in the Agro-Ecosystem

Despite the numerous benefits, the agricultural utilization of raw and composted wastes could also induce adverse impacts on the environment. This is particularly due to the types of toxic elements contained in these organic materials that might enter the food chain, since they may be taken up by plants from soil.

The concentration of heavy metals in compost is generally higher than the normal concentration in soil, so the possibility exists of metal accumulation when the compost is repeatedly applied (Zhang et al., 2006). On the other hand, little research is available from studies lasting decades and longer about the availability of metals applied as constituents of composted organic matter. Over a 10-year trial period, the most abundant metals in the uppermost soil horizon were copper (Cu), zinc (Zn) and lead (Pb). Cadmium (Cd) was the least plentiful, corresponding to the mean metal concentrations in the municipal solid waste compost applied (Businelli et al., 2009). This is supported by the opposite findings of Bergkvist et al. (2003) who found, during a period of 41 years, 92% of applied Cd was recovered in the topsoil in sludge treatment, indicating measurable losses by both downward movement and crop uptake.

Six-year consecutive applications of a swine compost resulted in significantly higher concentrations of Cu and Zn at 10–20 cm depth of the compost-amended soil, relative to the control, with an increase from 102.8 to 127.4 mg kg<sup>-1</sup> for Cu and from 111.9 to 165.7 mg kg<sup>-1</sup> for Zn (Zhao et al., 2006). On the other hand, Bartl et al. (2002) found that 32t ha<sup>-1</sup> of biowaste compost did not influence the total contents of Cd, manganese (Mn), molybdenum (Mo) or nickel (Ni) in soil, in 5 years. The total soil contents of Zn and Pb were significantly higher in soils with compost treatment than in the unfertilized soils.

From experiments longer than 10 years, it is possible to suggest that sludge and composted sludge showed a high accumulation of Zn, Cu and chromium (Cr), probably due to the notably higher concentration of these metals in the raw materials (Saviozzi et al., 1999; Kunito et al., 2001). The soil enzyme activities, dehydrogenase, urease and  $\beta$ -D-glucosidase were also found to be adversely affected by the metals derived from the addition of sewage sludge (Kunito et al., 2001).

Notwithstanding, the concentration of pollutants in composts may be reduced through the correct separation of organic wastes at source, which offers the opportunity of high-quality input material for aerobic biodegradation (Montemurro et al., 2009). For the long-term protection of the environment, it is also necessary to develop and implement other preliminary treatments for potentially polluting wastes. Erhart et al. (2008) showed that, after 10 years of application, the use of high-quality biowaste compost gives no variation in either total heavy metal concentrations or available fractions. In particular, with total applications of 95, 175 and 255 t biowaste compost ha<sup>-1</sup>, no heavy metal significant increase was measurable, except for Zn in the treatment with the highest application rate.

However, the environmental hazard is strictly linked to the mobility of metals, and to their concentration in the soil solution rather than to the total soil concentration. According to Businelli et al. (2009), metal mobilization is not an immediate process, but it involves various equilibria that control their adsorption and desorption. These behaviors depend on soil characteristics and climatic conditions and they involve biochemical processes responsible for the microbial degradation of autochthonous and compost-derived organic matter. This highlights the importance of long-term experimentation when studying the environmental fate of heavy metals, particularly in compost-amended soils on large scales (Businelli et al., 2009).

Within this context, according to Tittarelli et al. (2007), the main factors that affect the environmental behavior of heavy metals are: (i) cation exchange capacity, as an index of the soil capacity to adsorb and hold metal cations; (ii) humic substances, that can interact with heavy metals, forming complexes with different solubility, and consequently mobility, and (iii) the water and thermic regime of the soil, which affect the organic matter decomposition.

It can also be generally assumed that extractability and uptake of heavy metals decline as the soil pH becomes more alkaline, especially after repeated compost application. By contrast, low pH in the soil, caused by more than 60 years of rice straw compost applications, may have enhanced the concentration of metals in the water-soluble fraction (Takeda et al., 2005).

The aerobic composting processes increase the complexation of heavy metals in organic waste residues. In this condition the metals are strongly bound to the compost matrix and organic matter, limiting their solubility and potential bioavailability in soil (Smith, 2009). However, 10 years after municipal solid waste compost application, a share of the heavy metals was further re-mobilized in the soil profile, also leading to a decrease in the percentage distribution of organically-bound heavy metals with time. According to Businelli et al. (2009), this metal mobilization was primarily influenced by organic matter dynamics. On the contrary, Sukkariyah et al. (2005) reported that in aerobically digested sewage sludge-amended soils, the extractability of the heavy metals steadily declined over 17 years, despite a significant decline in organic matter concentration in the amended plots. This outcome showed that the mineralization of exogenous applied organic matter did not increase heavy metal availability due to loss in metal binding capacity associated with organic matter.

Monitoring concentrations of various heavy metals, their fractionation and the percentage contribution of each fraction to the total concentration are important analytical tools. In fact, in soil receiving long-term application of compost or other organic materials with relatively high concentrations of metals, these determinations can produce important information at the present time and help to determine what might happen in the future (Zhao et al., 2006; Businelli et al., 2009).

In any case, moderate compost doses applied in current agriculture will not cause any risk of toxicity for plants or animals, and therefore it is reasonable that application rates always have to be chosen on the basis of limited heavy metal loadings. However, no adverse effect on plant growth or excessive amounts of plant metal uptake were noted 17 to 19 years after biosolid application, despite the high application rate, equal to 210 t ha<sup>-1</sup>, containing concentrations of Cu and Zn that exceeded the pollutant concentration limits (Sukkariyah et al., 2005).

In general, heavy metal uptake by crops increases in leafy plants and it is higher in cereal leaves than in grain. Lettuce, for example, had higher assimilative capacity for Zn and Cd uptake than other nonleafy crops (Sukkariyah et al., 2005). Mantovi et al. (2005) reported that biosolid applications significantly increased the content of Zn and Cu in wheat grain and of only Cu in both sugarbeet roots and maize grain. Cadmium is one of the most significant potential contaminants of food supplies on arable lands and may limit sewage sludge suitability for soil amendment, because this organic material presents a large amount of this metal (Singh and Pandeya, 1998). Being relatively soluble in soils, it is readily taken up by crops and it is quite toxic to humans (Miller and Miller, 2000). This behavior is particularly evident at low soil pH. Cd solubility in equilibrium extracts of  $Ca(NO_3)_2$  increased, during the 41-year trial above mentioned, by a factor of 20 in the sludge treatment compared with the control (Bergkvist et al., 2003). This was reflected in the Cd concentration of the straw fraction in barley, which was almost doubled in the sewage sludge treatment, compared with the control. The grain fraction, on the other hand, showed no significant increase in Cd concentration.

It can be concluded that there is no tangible evidence demonstrating negative impacts of heavy metals applied to soil, particularly when high-quality compost was used for a long period. Composting processes also inherently reduce metal availability compared with other organic waste stabilization methods (Smith, 2009).

#### 3.3 N Pool Fate

Proper use of organic amendments requires the capacity to predict the release in the soil of inorganic nutrients from organic forms. Sikora and Szmidt (2001) suggested that a better understanding of coupled mineralization-nitrification is essential to manage the soil N pool with an environmentally sound application of amendments.

The C/N ratio, which is an important tool of amendment evaluation, cannot explain all differences in N mineralization, since organic materials with similar C/N ratios may mineralize different amounts of N. This behavior is probably due to other differences in their chemical compositions. Repeated long-term applications of organic amendments not only generally increase the size of the soil organic N pool, but also cause remarkable changes in soil characteristics, that influence N dynamics and can lead to a residual effect.

Habteselassie et al. (2006a) found an 89% increase in total soil N content after 5 years when dairywaste compost at 200 kg N ha<sup>-1</sup> was applied. Conversely, Zaman et al. (2004) observed, after 23 years of 240 kg Nha<sup>-1</sup> sewage sludge plus sawdust compost application, an increase in total N content of approximately 14%, as compared with a chemical fertilizer supply. The increase in total soil N after compost, biosolids or compost-plus-N was closely related to the build-up of organic matter in the soil over time and may be attributed to the direct effect of organic inputs (Mantovi et al., 2005; Ros et al., 2006a). After 7 years of application, the soil N concentration in the topsoil accounted for 33% of compost N applied (Sullivan et al., 2003). Antil et al. (2005) showed that the highest total soil N content increase, by 70% in respect to NPK mineral fertilizer, was found after 38 years of animal-manure liquid applications, as compared with solid animal-manure, cattle slurry and half cattle slurry plus straw. Moreover, in the plots after 13 years of cropping, compared with those kept fallow for the same period, there was a clear impact of N removal due to both the harvest and N losses, following enhanced mineralization due to tilling.

The N uptake by crops depends on several factors such as the N content and C/N ratio of the amendment, soil characteristics, climatic conditions and, obviously, on the plant's N requirements. Research on an average of 10 years by Hartl and Erhart (2005) pointed out that the ratio of N output by harvested plant parts to total N input in the source-separated organic waste compost treatments ranged from 3 to 7% of the applied N. This result was lower compared with mineral fertilization (11 to 15%), even if it was not reflected by a higher proportion of  $NO_3$ –N in the soil profile at the end of the growing season.

Data collected on N recovery from different composts, related to N uptake or yield, showed that the N effect of compost application does not generally exceed 15-20% of total N supply in the first year, while the residual compost N pool is mineralized at rates of 3-8% in following years. Continuous compost amendments and crop rotations with high nutrient demand may increase N mineralization (Amlinger et al., 2003). Eghball (2000) found that of the organic N applied to provide corn N requirements, 151 kg available N ha<sup>-1</sup> for an expected 9.4 t ha<sup>-1</sup> grain yield, about 11% was mineralized from composted beef cattle manure and 21% from non-composted manure during the succeeding growing season. A lower N availability from compost reflects the presence of stable N compounds.

The organic matter mineralization process increases the amount of ammonium and nitrate in the soil. However, the NO<sub>3</sub>–N is minimally adsorbed by the soil particles because it is very mobile and is susceptible to losses into ground- and surface waters by infiltrating water. Within this context, it is essential to remember that the synchronization of N supply with crop N demand, together with a proper application rate, is the best way to avoid the accumulation of soil mineral N, thus reducing the risk of NO<sub>3</sub>–N leaching (Montemurro and Maiorana, 2008).

The long-term application effects of raw and composted organic materials on nitrate leaching have been evaluated by several researchers, but the results of the studies are often discordant. In fact, Mallory and Griffin (2007) in a 13-year experiment found that, despite similar NH<sub>4</sub>–N inputs and rates of NH<sub>4</sub>–N consumption for manure and fertilizer N treatments, NO<sub>3</sub>– N accumulation was slower in the manure treatment. This is because the N from manure became available more slowly than fertilizer N. On the contrary, Basso and Ritchie (2005), in a 6-year maize–alfalfa rotation, observed the highest amount of NO<sub>3</sub>–N leaching of 681 kg ha<sup>-1</sup> in the manure treatment, followed by compost (390 kg ha<sup>-1</sup>), inorganic N (348 kg ha<sup>-1</sup>) and control (311 kg ha<sup>-1</sup>). They suggest that, although manure applications can be valuable for organic matter increase, attention needs to be given to the possible environmental impact without benefiting from yield increase.

From several experiments longer than 8 years conducted under different soil and climate conditions, it might be concluded that compost fertilization resulted in equal or lower NO<sub>3</sub>–N leaching losses to groundwater than corresponding mineral supply (Hartl et al., 2001; Hartl and Erhart, 2005; Erhart et al., 2007). These findings are observed even with higher amounts of amendments than used in practical farming. Nevens and Reheul (2003) reported that, at the end of a 4-year trial, the increase in residual soil NO<sub>3</sub>–N was smaller with vegetable-fruit-garden waste compost and compost plus cattle slurry applications, in respect to mineral fertilizer treatment.

Regarding the effect of the application rates, over a 5-year field trial with permanent rye, the treatments receiving 20 to 40 t ha<sup>-1</sup> of biowaste compost per single application showed smaller amplitudes in the soil NO<sub>3</sub>-N levels in fall than the treatments receiving 60 t  $ha^{-1}$  (Hartl et al., 2003). Other studies have noted that compost produced more nitrate than needed for plant use throughout a 5-year period. This is particularly evident at a high rate of application. The continuing mineralization of the organic material, even after harvest, led to postharvest nitrate production and a consequent downward movement (Habteselassie et al., 2006a). Furthermore, measurements of gross N transformation rates are important to properly understand N cycling in agricultural soils, where both productive and consumptive processes occur. Habteselassie et al. (2006b) reported that, by applying dairy-waste compost and liquid dairy-waste annually for 6 years, the mean gross N mineralization rates were about 5.7 and 2.9 mg N kg<sup>-1</sup> day<sup>-1</sup>. Conversely, the gross nitrification rates were 10.2 and 1.6 mg N kg<sup>-1</sup> day<sup>-1</sup>, respectively, and the net mineralization rates were less than 35% of gross rates. The ratio of gross nitrification to mineralization for dairy-waste compost-treated plots were more than 100%, indicating a large increase in nitrifying capacity of the soil following repeated applications, which is undesirable due to the known high mobility of nitrate.

Zaman et al. (2004) found that the long-term application of sewage sludge, composted with rice husk or sawdust, has showed a positive effect on gross and net rates of N mineralization, and net nitrification rates, probably due to high levels of microbial and enzyme activities. The same authors also found that chemical fertilizer-treated soils showed negative net N mineralization rates, which is indicative of N immobilization. These assessments are contradicted by Poudel et al. (2002), who observed that organic systems managed with cover crops and composted animal manure showed 112% greater potentially mineralizable N pools than conventional systems. They also showed N mineralization rates to be lower by 100% throughout the growing season, as compared with conventional systems. This reduction corresponds to a reduced risk for NO<sub>3</sub>–N leaching and groundwater pollution.

The valuation of the potentially mineralizable organic N pool for soils that have received organic amendments is important for estimating the capacity of these soils to provide N over time. For example, residual effects of 4-year compost and manure application resulted in 42 to 74% higher potentially mineralizable N compared with synthetic N fertilizer treatment (Ginting et al., 2003). Furthermore, when both organic amendments are added to soil and biological oxygen demand by decomposer microorganisms exceeds the supply, the anaerobic microenvironment necessary for denitrification is created and the intermediate product N<sub>2</sub>O gas is formed. This gas is a strong "greenhouse gas" and can also be emitted into the atmosphere (Epstein, 1997). The long-term application of stabilized organic materials at 300 kg N ha<sup>-1</sup> year<sup>-1</sup> significantly increased N<sub>2</sub>O emissions from 150 g N<sub>2</sub>O-N ha<sup>-1</sup> year<sup>-1</sup> in the control treatment soil to 856 g N<sub>2</sub>O–N  $ha^{-1}$  year<sup>-1</sup> in the organic ones (Meng et al., 2005). N<sub>2</sub>O production was also significantly higher in the 0-15 and 15-20 cm layers of the compost-treated soils than in the respective layers of the chemical fertilizertreated ones (Zaman et al., 2004). The higher N<sub>2</sub>O emissions from some compost-treated soils highlight the need for quantifying N2O emissions from agricultural land. This is particularly important where composts are used instead of chemical fertilizers (Sikora and Szmidt, 2001).

In summary, the repeated application of composted materials can enhance soil organic N content, storing it for mineralization in the following cropping seasons.

### **4** Conclusion

Only a few published studies have focused on the longterm effects of organic amendments on the soil-plant system for sustainable crop production. Therefore, our work attempted to address this issue by using the results from long-lasting fertility trials, providing a useful platform for future research.

The basic reflections of this review include:

- (i) long-term trials are the best indicators of both organic material utilization and sustainability. In fact, many effects, e.g. release of nutrients, carbon sequestration and possible build-up of toxic elements in the soil, evolve slowly, thus needing time to be tested. These effects are present only when repeated applications occurred;
- (ii) addition of exogenous organic matter to cropland can lead to an improvement in soil biological functions, even more than 15 years after spreading, depending on the quantity and quality of materials applied. For example, microbial biomass C increased up to 100% with high rate compost treatments, while enzymatic activity was enhanced by up to about 30% with sludge additions;
- (iii) several organic amendments' long-lasting application can enhance soil available potassium, extractable phosphorous and, particularly, organic carbon. Soil organic carbon increased by about 90%, in respect to unfertilized soil, and by more than 100% as compared with chemical fertilizers;
- (iv) regular addition of organic amendments increased soil physical fertility, mainly by improving aggregate stability and decreasing soil bulk density;
- (v) the best agronomic performance of compost is obtained with the highest rates and frequency of utilization. This is particularly evident if the amendment is combined with mineral fertilizers. Furthermore, there were residual effects because it is a slow-release nitrogen fertilizer;
- (vi) crop yield was enhanced by about 250%, with high rates of municipal solid waste compost repeated applications. There is also some evidence that stabilized organic amendments do not reduce the crop yield quality, but even enhance it;

- (vii) long-term organic amendment application can play a positive role in climate change mitigation by soil carbon sequestration, which in turn can reverse the process of soil degradation;
- (viii) agricultural utilization of raw and composted wastes could induce adverse environmental impacts. However, there is no tangible evidence demonstrating negative impacts of heavy metals applied to soil, particularly when high-quality compost was used for a long period. Composting processes also reduce metal availability;
- (ix) repeated application of composted materials can enhance soil organic N content, by up to 90%, storing it for mineralization in the following cropping seasons, often without raising the nitrate leaching to groundwater.

### References

- Abbott L.K., Murphy D.V. (2007) What is soil biological fertility? in: Abbott L.K., Murphy D.V. (Eds.), Soil biological fertility – A key to sustainable land use in agriculture, Springer, pp. 1–15, ISBN 978-1-4020-6619-1.
- Abiven S., Menasseri S., Chenu C. (2008) The effects of organic inputs over time on soil aggregate stability – A literature analysis, Soil Biol. Biochem. 41, 1–12.
- Albiach R., Canet R., Pomares F., Ingelmo F. (2001) Organic matter components and aggregate stability after the application of different amendments to a horticultural soil, Bioresource Technol. 76, 125–129.
- Amlinger F., Götz B., Dreher P., Geszti J., Weissteiner C. (2003) Nitrogen in biowaste and yard waste compost: dynamics of mobilisation and availability – a review, Eur. J. Soil Biol. 39, 107–116.
- Annabi M., Houot S., Francou C., Le Villio-Poitrenau M., Le Bissonnais Y. (2006) Improvement of aggregate stability after urban compost addition in a silty soil, in: Kraft E., Bidlingmaier W., de Bertoldi M., Diaz L.F., Barth J. (Eds.), Biological Waste Management, From Local to Global, Proceedings of the International Conference ORBIT 2006 /Eckhard Kraft...(Hrsg.), Weimar: Verlag ORBIT e.V., ISBN 3-935974-09-4.
- Antil R.S., Gerzabek M.H., Haberhauer G., Eder G. (2005) Long-term effects of cropped vs. fallow and fertilizer amendments on soil organic matter II. Nitrogen, J. Plant Nutr. Soil Sc. 168, 212–218.
- Baldock J.A., Nelson P.N. (2000) Soil organic matter, in: Sumner M.E. (Ed.), Handbook of soil science, CRC Press, Chap. 2.
- Barbarick K.A., Ippolito J.A. (2007) Nutrient assessment of a dryland wheat agroecosystem after 12 years of biosolids applications, Agron. J. 99, 715–722.
- Bartl B., Hartl W., Horak O. (2002) Long-term application of biowaste compost versus mineral fertilization: Effects on the

nutrient and heavy metal contents of soil and plant, J. Plant Nutr. Soil Sc. 165, 161–165.

- Basso B., Ritchie J.T. (2005) Impact of compost, manure and inorganic fertilizer on nitrate leaching and yield for a 6-year maize–alfalfa rotation in Michigan, Agr. Ecosys. Environ. 108, 329–341.
- Bastida F., Kandeler E., Hernández T., García C. (2008) Longterm effect of municipal solid waste amendment on microbial abundance and humus-associated enzyme activities under semiarid conditions, Microbial Ecol. 55, 651–661.
- Benbi D.K., Brar J.S. (2009) A 25-year record of carbon sequestration and soil properties in intensive agriculture, Agron. Sustain. Dev. 29, 257–265.
- Bergkvist P., Jarvis N., Berggren D., Carlgren K. (2003) Longterm effects of sewage sludge applications on soil properties, cadmium availability and distribution in arable soil, Agr. Ecosys. Environ. 97, 167–179.
- Bi L., Zhang B., Liu G., Li Z., Liu Y., Ye C., Yu X., Lai T., Zhang J., Yin J., Liang Y. (2008) Long-term effects of organic amendments on the rice yields for double rice cropping systems in subtropical China, Agr. Ecosys. Environ. DOI 10.1016/j.agee.2008.11.007.
- Blackshaw R.E., Molnar L.J., Larney F.J. (2005) Fertilizer, manure and compost effects on weed growth and competition with winter wheat in western Canada, Crop Prot. 24, 971–980.
- Businelli D., Massaccesi L., Said-Pullicino D., Gigliotti G. (2009) Long-term distribution, mobility and plant availability of compost-derived heavy metals in a landfill covering soil, Sci. Total Environ. 407, 1426–1435.
- Butler T.J., Muir J.P. (2006) Dairy manure compost improves soil and increases tall wheatgrass yield, Agron. J. 98, 1090–1096.
- Cherif H., Ayari F., Ouzari H., Marzorati M., Brusetti L., Jedidi N., Hassen A., Daffonchio D. (2009) Effects of municipal solid waste compost, farmyard manure and chemical fertilizers on wheat growth, soil composition and soil bacterial characteristics under Tunisian arid climate, Eur. J. Soil Biol. 1–8.
- Corbeels M., Hofman G., Cleemput O.V. (1999) Simulation of net N immobilization and mineralisation in substrateamended soils by the NCSOIL computer model, Biol. Fert. Soils 28, 422–430.
- Davies G., Lennartsson M. (2005) Organic vegetable production. A complete guide, Henry Doubleday Research Association, The Crowood Press Ldt., ISBN 1 86126 788 6.
- Dorado J., Zancada M.C., Almendros G., López-Fando C. (2003) Changes in soil properties and humic substances after long-term amendments with manure and crop residues in dryland farming system, J. Plant Nutr. Soil Sc. 166, 31–38.
- Durán Zuazo V.H., Rodríguez Pleguezuelo C.R. (2008) Soilerosion and runoff prevention by plant covers. A review, Agron. Sustain. Dev. 28, 65–86.
- Düring R.-A., Gäth S. (2002) Utilization of municipal organic wastes in agriculture: where do we stand, where will we go? J. Plant Nutr. Soil Sc. 165, 544–556.
- EC Council Regulation N. 834 (2007) on organic production and labelling of organic products and repealing Regulation (EEC) N. 2092/91, Official Journal of the European Union.

- Edmeades D.C. (2003) The long-term effects of manures and fertilisers on soil productivity and quality: a review, Nutr. Cycl. Agroecosys. 66, 165–180.
- Eghball B. (2000) Nitrogen mineralization from field-applied beef cattle feedlot manure or compost, Soil Sci. Soc. Am. J. 64, 2024–2030.
- Eghball B. (2002) Soil properties as influenced by phosphorusand nitrogen-based manure and compost applications, Agron. J. 94, 128–135.
- Eghball B., Ginting D., Gilley J.E. (2004) Residual effects of manure and compost applications on corn production and soil properties, Agron. J. 96, 442–447.
- Elfstrand S., Hedlund K., Mårtensson A. (2007) Soil enzyme activities, microbial community composition and function after 47 years of continuous green manuring, Appl. Soil Ecol. 35, 610–621.
- Epstein E. (1997) The science of composting, CRC Press LLC.
- Erhart E., Feichtinger F., Hartl W. (2007) Nitrogen leaching losses under crops fertilized with biowaste compost compared with mineral fertilization, J. Plant Nutr. Soil Sc. 170, 608–614.
- Erhart E., Hartl W., Putz B. (2005) Biowaste compost affects yield, nitrogen supply during the vegetation period and crop quality of agricultural crops, Eur. J. Agron. 23, 305–314.
- Erhart E., Hartl W., Putz B. (2008) Total soil heavy-metal concentrations and mobile fractions after 10 years of biowastecompost fertilization, J. Plant Nutr. Soil Sc. 171, 378–383.
- Feller C., Bernoux M. (2008) Historical advances in the study of global terrestrial soil organic carbon sequestration, Waste Manage. 28, 734–740.
- García-Gil J.C., Ceppi S.B., Velasco M.I., Polo A., Senesi N. (2004) Long-term effects of amendment with municipal solid waste compost on the elemental and acidic functional group composition and pH-buffer capacity of soil humic acids, Geoderma 121, 135–142.
- Ginting D., Kessavalou A., Eghball B., Doran J.W. (2003) Greenhouse gas emissions and soil indicators four years after manure and compost applications, J. Environ. Qual. 32, 23–32.
- Habteselassie M.Y., Miller B.E., Thacker S.G., Stark J.M., Norton J.M. (2006a) Soil nitrogen and nutrient dynamics after repeated application of treated dairy-waste, Soil Sci. Soc. Am. J. 70, 1328–1337.
- Habteselassie M.Y., Stark J.M., Miller B.E., Thacker S.G., Norton J.M. (2006b) Gross nitrogen transformations in an agricultural soil after repeated dairy-waste application, Soil Sci. Soc. Am. J. 70, 1338–1348.
- Hartl W., Erhart E. (2005) Crop nitrogen recovery and soil nitrogen dynamics in a 10-year field experiment with biowaste compost, J. Plant Nutr. Soil Sc. 168, 781–788.
- Hartl W., Erhart E., Schott W. (2001) Nitrogen dynamics in compost-amended systems, in: Federal Ministry of Agriculture, Forestry, Environment and Water Management, Austria, and European Communities, Applying compost benefits and needs, Seminar Proceedings Brussels, 22–23 November 2001, ISBN 3-902 338-26-1.
- Hartl W., Putz B., Erhart E. (2003) Influence of rates and timing of biowaste compost application on rye yield and soil nitrate levels, Eur. J. Soil Biol. 39, 129–139.
- He Z., Yang X., Kahn B.A., Stoffella P.J., Calvert D.V. (2001) Plant nutrition benefits of phosphorus, potassium, calcium,

magnesium, and micronutrients from compost utilization, in: Stoffella P.J., Kahn B.A. (Eds.), CRC Press LLC, Compost utilization in horticultural cropping systems, pp. 307–317, ISBN 1-56670-460-X.

- Insam H., de Bertoldi M. (2007) Microbiology of the composting process, in: Diaz L.F., de Bertoldi M., Bidlingmaier W., Stentiford E. (Eds.), Compost Science and Technology, Waste management series 8, Elsevier Ltd., pp. 26–45, ISBN-13: 978-0-08-043960-0.
- Ismail I., Blevins R.L., Frye W.W. (1994) Long-term no-tillage effects on soil properties and continuous corn yields, Soil Sci. Soc. Am. J. 58, 193–198.
- Johnson J.M.F., Reicosky D.C., Allmaras R.R., Sauer T.J., Venterea R.T., Dell C.J. (2005) Greenhouse gas contributions and mitigation potential of agriculture in the central USA, Soil Till. Res. 83, 73–94.
- Kaur T., Brar B.S., Dhillon N.S. (2008) Soil organic matter dynamics as affected by long-term use of organic and inorganic fertilizers under maize–wheat cropping system, Nutr. Cycl. Agroecosys. 81, 59–69.
- Komatsuzaki M. (2004) Use of cover crops in upland fields (in Japanese), Jpn J. Farm Work Res. 39, 157–163.
- Komatsuzaki M., Ohta H. (2007) Soil management practices for sustainable agro-ecosystems, Sustain. Sci. 2, 103–120.
- Kukal S.S., Rehana-Rasool, Benbi D.K. (2009) Soil organic carbon sequestration in relation to organic and inorganic fertilization in rice–wheat and maize–wheat systems, Soil Till. Res. 102, 87–92.
- Kunito T., Saeki K., Goto S., Hayashi H., Oyaizu H., Matsumoto S. (2001) Copper and zinc fractions affecting microorganisms in long-term sludge amended soils, Bioresource Technol. 79, 135–146.
- Lal R. (2004) Soil carbon sequestration to mitigate climate change, Geoderma 123, 1–22.
- Lal R. (2006) Enhancing crop yields in the developing countries through restoration of the soil organic carbon pool in agricultural lands, Land Degrad. Dev. 17, 197–209.
- Lal R. (2007) Carbon management in agricultural soils, Mitigation and Adaptation Strategies for Global Change 12, 303–322.
- Lal R. (2008) Soils and sustainable agriculture. A review, Agron. Sustain. Dev. 28, 57–64.
- Leon M.C.C., Stone A., Dick R.P. (2006) Organic soil amendments: Impacts on snap bean common root rot (Aphanomyes euteiches) and soil quality, Appl. Soil Ecol. 31, 199–210.
- Leroy B.L.M.M., Bommele L., Reheul D., Moens M., De Neve S. (2007) The application of vegetable, fruit and garden waste (VFG) compost in addition to cattle slurry in a silage maize monoculture: Effects on soil fauna and yield, Eur. J. Soil Biol. 43, 91–100.
- Lichtfouse E., Navarrete M., Debaeke P., Souchère V., Alberola C., Ménassieu J. (2009) Agronomy for sustainable agriculture. A review, Agron. Sustain. Dev. 29, 1–6.
- Madejón E., López R., Murillo J.M., Cabrera F. (2001) Agricultural use of three (sugar-beet) vinasse composts: effect on crops and chemical properties of a Cambisol soil in the Guadalquivir river valley (SW Spain), Agr. Ecosys. Environ. 84, 55–65.
- Mallory E.B., Griffin T.S. (2007) Impacts of soil amendment history on nitrogen availability from manure and fertilizer, Soil Sci. Soc. Am. J. 71, 964–973.

- Mandal B., Majumder B., Bandyopadhyay P.K., Hazra G.C., Gangopadhyay A., Samantaray R.N., Mishra A.K., Chaudhury J., Saha M.N., Kundu S. (2007) The potential of cropping systems and soil amendments for carbon sequestration in soils under long-term experiments in subtropical India, Global Change Biol. 13, 357–369.
- Mantovi P., Baldoni G., Toderi G. (2005) Reuse of liquid, dewatered, and composted sewage sludge on agricultural land: effects of long-term application on soil and crop, Water Res. 39, 289–296.
- Mats J., Lennart T. (1999) Impact of organic residues on soil microorganisms, in: Bidlingmaier W., de Bertoldi M., Diaz L.F., Papadimitriou E.K. (Eds.), Organic Recovery and Biological Treatment, Proceedings of the International Conference ORBIT 99 on Biological Treatment of Waste and the Environment/Werner Bidlingmaier (Hrsg.), Berlin: Rhombos-Verl, ISBN 3-930894-20-3.
- Mayhew L. (2004) Humic Substances in Biological Agriculture, Acres USA 34.
- Meng L., Ding W., Cai Z. (2005) Long-term application of organic manure and nitrogen fertilizer on N<sub>2</sub>O emissions, soil quality and crop production in a sandy loam soil, Soil Biol. Biochem. 37, 2037–2045.
- Miller D.M., Miller W.P. (2000) Land application of wastes, in: Sumner M. (Ed.), Handbook of soil science, CRC Press, Chap. 9.
- Monaco S., Hatch D. J., Sacco D., Bertora C., Grignani C. (2008) Changes in chemical and biochemical soil properties induced by 11-yr repeated additions of different organic materials in maize-based forage systems, Soil Biol. Biochem. 40, 608–615.
- Mondini C., Sequi P. (2008) Implication of soil C sequestration on sustainable agriculture and environment, Waste Manage. 28, 678–684.
- Monnier G. (1965) Action des matières organiques sur la stabilité structurale des sols, Thèse de la faculté des sciences de Paris, 140 p.
- Montemurro F. (2009) Different nitrogen fertilization sources, soil tillage, and crop rotations in winter wheat: effect on yield, quality, and nitrogen utilization, J. Plant Nutr. 32, 1–18.
- Montemurro F., Maiorana M. (2008) Organic Fertilization as Resource for a Sustainable Agriculture, Fertilizers: Properties, Applications and Effects, in: Elsworth L.R. et al. (Eds.), pp. 123–146, ISBN 978-1-60456-483-9.
- Montemurro F., Convertini G., Ferri D. (2004) Mill wastewater and olive pomace compost as amendments for rye-grass, Agronomie 24, 481–486.
- Montemurro F., Diacono M., Vitti C., Debiase G. (2009) Biodegradation of olive husk mixed with other agricultural wastes, Bioresource Technol. 100, 2969–2974.
- Montemurro F., Maiorana M., Convertini G., Ferri D. (2006) Compost organic amendments in fodder crops: effects on yield, nitrogen utilization and soil characteristics, Compost Sci. Util. 14, 114–123.
- Montemurro F., Maiorana M., Convertini G., Ferri D. (2007) Alternative sugar beet production using shallow tillage and municipal solid waste fertilizer, Agron. Sustain. Dev. 27, 129–137.
- Moral R., Paredes C., Bustamante M.A., Marhuenda-Egea F., Bernal M.P. (2009) Utilisation of manure composts by high-

value crops: Safety and environmental challenges, Bioresource Technol. 100, 5454–5460.

- Murphy D.V., Stockdale E.A., Brookes P.C., Goulding K.W.T. (2007) Impact of microorganisms on chemical transformation in soil, in: Abbott L.K., Murphy D.V. (Eds.), Soil biological fertility – A key to sustainable land use in agriculture, Springer, pp. 37–59, ISBN 978-1-4020-6619-1.
- Nardi S., Morari F., Berti A., Tosoni M., Giardini L. (2004) Soil organic matter properties after 40 years of different use of organic and mineral fertilisers, Eur. J. Agron. 21, 357–367.
- Nevens F., Reheul D. (2003) The application of vegetable, fruit and garden waste (VFG) compost in addition to cattle slurry in a silage maize monoculture: nitrogen availability and use, Eur. J. Agron. 19, 189–203.
- Poudel D.D., Horwath W.R., Lanini W.T., Temple S.R., van Bruggen A.H.C. (2002) Comparison of soil N availability and leaching potential, crop yields and weeds in organic, low-input and conventional farming systems in northern California, Agr. Ecosys. Environ. 90, 125–137.
- Procházková B., Hrubý J., Dovrtìl J., Dostál O. (2003) Effects of different organic amendment on winter wheat yields under long-term continuous cropping, Plant Soil Environ. 49, 433–438.
- Ros M., Klammer S., Knapp B., Aichberger K., Insam H. (2006a) Long-term effects of compost amendment of soil on functional and structural diversity and microbial activity, Soil Use Manage. 22, 209–218.
- Ros M., Pascual J.A., Garcia C., Hernandez M.T., Insam H. (2006b) Hydrolase activities, microbial biomass and bacterial community in a soil after long-term amendment with different composts, Soil Biol. Biochem. 38, 3443–3452.
- Rothamsted Research (2006) Guide to the classical and other long-term experiments, datasets and sample archive, Lawes Agricultural Trust Co. Ltd., ISBN 0951445693, web: http: www.rothamsted.ac.uk.
- Sánchez-Monedero M.A., Cayuela M.L., Mondini C., Serramiá N., Roig A. (2008) Potential of olive mill wastes for soil C sequestration, Waste Manage. 28, 767–773.
- Saviozzi A., Biasci A., Riffaldi R., Levi-Minzi R. (1999) Longterm effects of farmyard manure and sewage sludge on some soil biochemical characteristics, Biol. Fert. Soils 30, 100–106.
- Sequi P. (1989) Le funzioni agronomiche della sostanza organica, in: Patron (Ed.), Chimica del suolo, Bologna, Italy, pp. 279–292.
- Shindo H., Hirahara O., Yoshida M., Yamamoto A. (2006) Effect of continuous compost application on humus composition and nitrogen fertility of soils in a field subjected to double cropping, Biol. Fert. Soils 42, 437–442.
- Sikora L.J., Szmidt R.A.K. (2001) Nitrogen sources, mineralization rates, and nitrogen nutrition benefits to plants from composts, in: Stoffella P.J., Kahn B.A. (Eds.), CRC Press LLC, Compost utilization in horticultural cropping systems, pp. 287–302, ISBN 1-56670-460-X.
- Singh A.K., Pandeya S. B. (1998) Modelling uptake of cadmium by plants in sludge-treated soils, Bioresource Technol. 66, 51–58.
- Sleutel S., De Neve S., Németh T., Tóth T., Hofman G. (2006) Effect of manure and fertilizer application on the distribution

of organic carbon in different soil fractions in long-term field experiments, Eur. J. Agron. 25, 280–288.

- Smith S.R. (2009) A critical review of the bioavailability and impacts of heavy metals in municipal solid waste composts compared to sewage sludge, Environ. Int. 35, 142–156.
- Sodhi G.P.S., Beri V., Benbi D.K. (2009) Soil aggregation and distribution of carbon and nitrogen in different fractions under long-term application of compost in rice–wheat system, Soil Till. Res., DOI 10.1016/j.still.2008.12.005.
- Sukkariyah B.F., Evanylo G., Zelazny L., Chaney R.L. (2005) Cadmium, Copper, Nickel, and Zinc availability in a biosolids-amended piedmont soil years after application, J. Environ. Qual. 34, 2255–2262.
- Sullivan D.M., Bary A.I., Nartea T.J., Myrhe E.A., Cogger C.G., Fransen S.C. (2003) Nitrogen availability seven years after a high-rate food waste compost application, Compost Sci. Util. 11, 265–275.
- Takeda A., Tsukada H., Nanzyo M., Takaku Y., Uemura T., Hisamatsu S., Inaba J. (2005) Effect of long-term fertilizer application on the concentration and solubility of major and trace elements in a cultivated Andisol, Soil Sci. Plant Nutr. 51, 251–260.
- Tejada M., Garcia C., Gonzalez J.L., Hernandez M.T. (2006) Organic amendment based on fresh and composted beet vinasse: influence on soil properties and wheat yield, Soil Sci. Soc. Am. J. 70, 900–908.
- Tejada M., Gonzalez J.L., García-Martínez A.M., Parrado J. (2008) Application of a green manure and green manure composted with beet vinasse on soil restoration: Effects on soil properties, Bioresource Technol. 99, 4949–4957.
- Tejada M., Hernandez M.T., Garcia C. (2009) Soil restoration using composted plant residues: Effects on soil properties, Soil Till. Res. 102, 109–117.
- Tilman D., Cassman K.G., Matson P.A., Naylor R., Polasky S. (2002) Agricultural sustainability and intensive production practices, Nature 418, 671–677.
- Tittarelli F., Petruzzelli G., Pezzarossa B., Civilini M., Benedetti A., Sequi P. (2007) Quality and agronomic use of compost, in: Diaz L.F., de Bertoldi M., Bidlingmaier W., Stentiford E. (Eds.),

Compost science and technology, Waste management series 8, Elsevier Ltd., pp. 119–145, ISBN-13: 978-0-08-043960-0.

- Triberti L., Nastri A., Giordani G., Comellini F., Baldoni G., Toderi G. (2008) Can mineral and organic fertilization help sequestrate carbon dioxide in cropland? Eur. J. Agron. 29, 13–20.
- Van-Camp L., Bujarrabal B., Gentile A.-R., Jones R.J.A., Montanarella L., Olazabal C., Selvaradjou S.-K. (2004). Reports of the Technical Working Groups Established under the Thematic Strategy for Soil Protection, EUR 21319 EN/3, 872 p., Office for Official Publications of the European Communities, Luxembourg.
- Watson C.A., Atkinson D., Gosling P., Jackson L.R., Rayns F.W. (2002). Managing soil fertility in organic farming systems, Soil Use Manage. 18, 239–247.
- Weber J., Karczewska A., Drozd J., Licznar M., Licznar S., Jamroz E., Kocowicz A. (2007) Agricultural and ecological aspects of a sandy soil as affected by the application of municipal solid waste composts, Soil Biol. Biochem. 39, 1294–1302.
- Westerman P.W., Bicudo J.R. (2005) Management considerations for organic waste use in agriculture, Bioresource Technol. 96, 215–221.
- Whalen J.K., Chang C. (2002) Macroaggregate characteristics in cultivated soils after 25 annual manure applications, Soil Sci. Soc. Am. J. 66, 1637–1647.
- Zaman M., Matsushima M., Chang S.X., Inubushi K., Nguyen L., Goto S., Kaneko F., Yoneyama T. (2004) Nitrogen mineralization, N<sub>2</sub>O production and soil microbiological properties as affected by long-term applications of sewage sludge composts, Biol. Fert. Soils 40, 101–109.
- Zhang M., Heaney D., Henriquez B., Solberg E., Bittner E. (2006) A four-year study on influence of biosolids/MSW cocompost application in less productive soils in Alberta: nutrient dynamics, Compost Sci. Util. 14, 68–80.
- Zhao B., Maeda M., Zhang J., Zhu A., Ozaki Y. (2006) Accumulation and chemical fractionation of heavy metals in Andisols after a different, 6-year fertilization management, Environ. Sci. Pollut. R. 13, 90–97.

# **Tillage Management Effects on Pesticide Fate in Soils**

Lionel Alletto, Yves Coquet, Pierre Benoit, Djilali Heddadj, and Enrique Barriuso

### Contents

1	Intro	duction	788
2	Inter	ception – Retention	790
	2.1	Mulch Effect	790
	2.2	Organic Carbon Content Effect	797
	2.3	Soil pH Effect	799
	2.4	Other Effects	799
	2.5	Sorption of Metabolites	799
	2.6	Summary and Recommendations	799
3	Degra	adation	799
	3.1	Pesticide Degradation in the Mulch	800
	3.2	Pesticide Degradation in Soil	800
	3.3	Summary and Recommendations	807
4	Trans	sfer of Pesticides	807
	4.1	Volatilisation	807
	4.2	Leaching	808
	4.3	Runoff	814
	4.4	Summary and Recommendations	824
5	Conc	lusion	824
Re	ferenc	es	825

Abstract Reducing tillage intensity through the implementation of conservation practices is a way to reach a more sustainable agriculture. Reducing tillage is indeed an efficient way to control soil erosion and to decrease production costs. Nonetheless, the environmental impact of reduced tillage is not well known because conservation techniques may induce strong changes in soil physicochemical properties and biological activity. Knowledge on the fate of applied pesticides under conservation practices is particularly important from this point of view. We review here the advances in the understanding, quantification and prediction of the effects of tillage on pesticide fate in soils. We found the following major points: (1) for most dissipation processes such as retention, degradation and transfer, results of pesticide behaviour studies in soils are highly variable and sometimes contradictory. This variability is partially explained by the multiplicity of processes and contributive factors, by the variety of their interactions, and by their complex temporal and spatial dynamics. In addition, the lack of a thorough description of tillage systems and sampling strategy in most reports hampers any comprehensive interpretation of this variability. (2) Implementation of conservation tillage induces an increase in organic matter content at the soil surface and its gradual decrease with depth. This, in turn, leads to an increase in pesticide retention in the topsoil layer. (3) Increasing retention of pesticides in the topsoil layer under conservation tillage decreases the availability of the pesticides for biological degradation. This competition between retention and degradation leads to a higher persistence of pesticides in soils, though this persistence can be partially compensated for by a more intensive microbial activity under conservation tillage. (4) Despite strong

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_35,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Alletto et al., Agron. Sustain. Dev. 30 (2010) 367–400. DOI: 10.1051/agro:2009018

L. Alletto (🖂)

Université de Toulouse – École d'ingénieurs de Purpan, UMR 1248 AGIR INRA-INP, 75 voie du TOEC, BP 57 611, 31076 Toulouse Cedex 3, France e-mail: lionel.alletto@purpan.fr

changes in soil physical properties under conservation tillage, pesticide transfer is more influenced by initial soil conditions and climatic conditions than by tillage. Conservation tillage systems such as no-tillage improve macropore connectivity, which in turn increases pesticide leaching. We conclude that more knowledge is needed to fully understand the temporal and spatial dynamics of pesticide in soil, especially preferential flows, in order to improve the assessment of pesticide risks, and their relation to tillage management.

**Keywords** Conventional tillage • Conservation tillage • Herbicide • Retention • Degradation • Transport • Soil carbon

#### **1** Introduction

Tillage can be defined as any mechanical operation on the soil and crop residues that aims at providing a suitable seedbed where crop seeds are sown. Soil preparation has always been an important component of traditional agriculture. Since 6000 BC, a wide variety of farming tools have been developed, from the simple digging stick to the paddle-shaped spade that could be pulled by humans or animals (Lal et al., 2007). In the 18th century, the mouldboard plough, a curved board that cuts and rotates the soil at a significant working depth, of more than 20 cm, was introduced in England. This tool was particularly efficient in incorporating organic residues, fertilisers and lime, in controlling weeds and helping the growing process. In the early part of the 19th century, the "initial" model of the mouldboard plough received several improvements to reach its "standardised" version by 1870, similar to current models (Hanson, 2006; Lal et al., 2007).

In the United States, between 1910 and 1930, the use of the mouldboard plough widely expanded, especially in the Great Plains of the Middle West. In Europe, its use started to be significant after 1945. These changes in tillage practices were accompanied by significant progresses in seed selection and by the development of the use of chemicals for plant protection, allowing a significant increase in crop yields. For example, wheat yields in France increased from 0.8-1 tons per hectare in 1945 to more than 8 tons in 1985. Because of its success, the mouldboard plough became a centrepiece of traditional agriculture.

In parallel with this growth of agricultural production, the widespread use of the mouldboard plough led to severe problems of soil and environmental degradation. One of the most well-known environmental problems was the drastic increase in wind erosion known as the "Dust Bowl" in 1930 in the US Great Plains (Lal et al., 2007; Masutti, 2004). Faced with this major problem, depriving some areas of any production capacity, the American reaction was swift. In 1935, the Soil Conservation Service - now called the Natural Resources Conservation Service - was created within the United States Department of Agriculture. Many important research programmes and communication efforts aimed at promoting the adoption of soil conservation techniques, grouped under the term "conservation tillage". These techniques are defined as any tillage and planting system that leaves at least 30% of the soil surface covered by crop residue after planting to reduce soil erosion by water, or at least 1.1 tons of crop residue/ha to reduce soil erosion by wind. A broad spectrum of farming methods can thus be classified as conservation techniques: from tillage techniques with a high working depth, such as sub-soiling, to reduced tillage of 0-15 cm or direct drilling techniques without any preliminary soil tillage, called "no-tillage". This last type of tillage technique remained less developed until 1940 with the discovery of hormonal herbicides such as 2,4-dichlorophenoxy acetic acid (2,4D) that allowed farmers to control weeds without ploughing. Fairly quickly, these North American techniques thrived in the South American countries, particularly in Brazil where water erosion was significant (Bernoux et al., 2006), and in Argentina and Chile. Tillage practices and their codes used in this review are reported in Table 1.

In Europe, because soil erosion was not as great as in the US, the interest of farmers in conservation tillage was limited. In the years 1970–1980, some attempts to develop these techniques were faced with decreases in crop yields, while, at the same time, farmers were encouraged to produce more to earn more. At the beginning of the 1990s, with the Rio conference (1992) which laid down the foundations for a sustainable agriculture, the concerns moved towards considering the vulnerability of water, soil and landscape resources. Moreover, the emergence of agricultural product surpluses led to lower prices. The control by farmers of their income more than ever implied lower production costs and a further increase in productivity, especially

Code	Tillage practice	$\%$ soil surface covered by crop residues $^{\dagger}$	Agricultural tool
СТ	Conventional tillage	<30% (0–10%)	Mouldboard plough
			Disk plough
			Spading machine
CnT	Conservation tillage		
MT <sub>Dk</sub>	Mulch tillage	>30% (30-50%)	Disk harrow
MT <sub>Ch</sub>	Mulch tillage	>30% (30-50%)	Chisel plough, harrow
SS	Sub-soiling	>30%(30-60%)	Sub-soiler, deep ripper, paratill
RT	Ridge tillage	>30%(40-60%)	Ridger
ST	Strip tillage	>30%(40-60%)	Strip-till
NT	No-tillage (or direct drilling)	>50% (depending on crop residues)	Seeding drill (with disc openers)

Table 1 Tillage practice types used in this review article

<sup>†</sup> (CTIC, 2006).



**Fig. 1** Evolution of cultivated surfaces without ploughing in France between 1994 and 2006. Whatever the crop, the trend is an increase of surfaces cultivated without ploughing and winter

crops are more often sown without ploughing than spring crops (Agreste, 2004, 2008)

as the size of farms expanded. Under this economic context, soil conservation tillage, deemed to be faster and less costly in time and energy than conventional tillage, finally started to become attractive.

Progressively, surfaces sown in France without ploughing increased for all crops (Fig. 1). Spring crops are, however, more frequently sown after ploughing than winter crops. Moreover, the development of conservation techniques varies according to the regions. For example, in the south-west part of France – Aquitaine and Midi-Pyrénées regions –, more than 75% of the wheat surfaces were not ploughed in 2006 against 50% in the whole country (Agreste, 2008).

One of the main roles of tillage is to provide an efficient control of weeds. Indeed, tillage influences weed populations by the combined effects of mechanical destruction of weed seedlings and by changing the vertical distribution of weed seeds in soil (Peigné et al., 2007). It also changes the soil climatic conditions which control weed dormancy, germination and growth. Reducing tillage intensity generally tends to increase the concentration of weeds in the topsoil (Moonen and Barberi, 2004; Torresen et al., 2003; Vasileiadis et al., 2007) and is very often associated with an increase in herbicide use (Fig. 2).

Although many studies have been conducted, mainly on the North American continent, the environmental fate of pesticides under conservation tillage presents many contradictions and remains, finally, poorly understood (Aubertot et al., 2005; Réal et al., 2007). Publication of articles concerning the effects of tillage practices on pesticide fate in soils, water and



**Fig. 2** Mean number of herbicide treatments applied in 2006 for several crops sown after ploughing (conventional tillage) or without ploughing (conservation tillage). Note the systematic

increase of herbicide treatments with the implementation of conservation tillage (Agreste, 2004, 2008)

air started slowly during the 1970s and showed a rapid increase at the end of the 1980s followed by a peak rate of publication at the end of the 1990s (Fig. 3a). Onethird of the published articles were found in one journal, and the first three journals were from the US, illustrating the great interest of US scientists in this subject (Fig. 3b).

Reducing tillage intensity leads to significant and complex changes in soil physical, chemical and biological properties, most often interrelated with each other, thus affecting the fate of the applied pesticides. A first review, conducted by Locke and Bryson (1997), dealt with herbicide interactions in soils under plant residuemanaged farming systems. Following their work, we sought to assess the current state of knowledge of the effects of tillage on the processes involved in the fate of pesticides in soils cultivated with arable crops. Figure 4 summarises the main changes in soil properties due to tillage practices and their effects on the mechanisms involved in pesticide fate. As shown in this figure, understanding the effects of tillage on pesticide fate in soils implies a comprehensive evaluation of all the interactions between the different dissipation processes and of all the different soil factors affected by tillage operations. Some of these interactions are now well known, but most of them are still poorly understood. In this review, for each of these processes, we paid particular attention to the understanding of the mechanisms involved, their relative importance and the origin of the contradictory results found. The lists of the molecules, mainly herbicides, mentioned in this review are shown in Table 2.

#### 2 Interception – Retention

Pesticide interception depends on several factors such as the type of application (pre- or post-emergence), the presence of a crop and, if any, plant growth stage and species, the presence of a mulch, etc. Because interception of pesticides by plant foliage and its effect on the environment have not been widely studied (Reddy and Locke, 1996), the subject developed in this part focuses mainly on the interception and retention by the mulch and by the soil. In some cases, due to the difficulty of formally separating interception and retention, these processes are discussed simultaneously.

### 2.1 Mulch Effect

The accumulation of organic residues on soil surface (mulch) in conservation tillage generally leads to an increase in the interception of applied pesticides, more particularly for apolar pesticides or pesticides with a low polarity (Locke, 1992; Novak et al., 1996; Reddy and Locke, 1998; Reddy et al., 1997a, b; Zablotowicz et al., 2000). This interception depends on the amount and type of crop residues and, for example, with a proportion  $\geq 30\%$  of the soil surface covered, it was found to range from 40 to 70% of the applied dose for several preemergence herbicides (Banks and Robinson, 1982; Ghadiri et al., 1984; Isensee and Sadeghi, 1994; Sadeghi and Isensee, 1997). In terms of mass of residues, for a quantity >4.5 t ha<sup>-1</sup>, the interception of



**Fig. 3** Articles published on the effects of tillage practices on pesticide fate in soils, water and air. (a) Time series of articles found per year; and (b) major source journals found in (a). The

plot includes the top 8 journals representing 115 articles out of the 144 references cited in the review

acetochlor, alachlor and metolachlor was higher than 90% of the applied doses (Banks and Robinson, 1986). Retention processes are closely associated with the interception of pesticides. Indeed, crop residues can have sorption capacities 10 to 60 times higher than soil (Boyd et al., 1990; Reddy et al., 1995b) and can significantly modify the (bio)availability and the migration of pesticides in soil. As a result, a loss of efficacy of some pesticides has been observed (Erbach and Lovely, 1975; Mills et al., 1989; Shelton et al., 1998) which can lead to an increase in the applied doses (Shelton et al., 1998; Worsham, 1991). By penetrating inside crop residues, most of the molecules are physically entrapped within cell wall structures, such as cellulose microfibrils embedded in a ligninhemicellulose matrix (Dao, 1991) and, most often, lose their activity.

The nature and decomposition degree of crop residues both influence interception and retention of pesticides, but in contrasted ways depending on studies. For example, hairy vetch (*Vicia villosa* Roth) residues had a higher retention for chlorimuron than rye (*Secale cereale* L.) residues (Reddy et al., 1995) and a higher retention for fluometuron than wheat (*Triticum aestivum* L.) residues (Gaston et al., 2001) (Table 3). This greater sorption by vetch than rye or wheat residues may be related to differences in physical state, that offered a greater surface area for



**Fig. 4** Relationships between the processes conditioning the fate of pesticides in soils, water and air and the soil factors modified by tillage operations

herbicide sorption in the case of the vetch, and in composition of the residues that contained less cellulose and more amino acids (Gaston et al., 2001). Sigua et al. (1993) indicated that interception of atrazine was enhanced with fresh maize residues, due to a combination of a greater hydrophobicity and a higher sorption capacity of the fresh compared with the aged maize residues. In contrast, with metribuzin (Dao, 1991), chlorimuron (Reddy et al., 1995b) and cyanazine (Reddy et al., 1997a), interception was higher with aged residues. In these cases, the increase in sorption due to aging was attributed to both changes in the physical state and in the chemical composition of the crop residues. On one hand, physical alterations of the residues increase their external surface area for herbicide sorption compared with fresh residues. On the other hand, the chemical evolution of the residues during degradation leads to an increase

in the lignin/cellulose ratio, resulting in an increase in the sorption of chlorimuron and metribuzin (Dao, 1991; Reddy et al., 1995b).

Interception is not only influenced by the amount of mulch. It is also conditioned by the formulation of the molecules. Compared with the commercial formulation, the use of microencapsulated or granular forms of alachlor led to a decrease in its interception and, finally, the herbicide reached the soil surface more rapidly (Johnson et al., 1989; Sadeghi et al., 1998).

Once intercepted by the mulch, the molecule can be washed off from the crop residues to reach the soil surface. This wash-off depends on the timing between pesticide treatment and the first rainfall and its intensity. On maize residues, Martin et al. (1978) reported a range of 30 to 60% wash-off by the first 5 mm of water for alachlor, atrazine, cyanazine and propachlor, which was equivalent to the wash-off

Common name	Nature	Chemical name							
Acetochlor	Herbicide	2'-ethyl-6'-methyl-N-(ethoxymethyl)-2-chloroacetylanilide							
Acifluorfen	Herbicide	5-[2-chloro-4-(trifluoromethyl)phenoxy]-2-nitrobenzoic acid							
Alachlor	Herbicide	2-chloro- $N$ -(2, 6-diethylphenyl)- $N$ -(methoxymethyl)acetamide							
Atrazine	Herbicide	(6-chloro-N-ethyl)-N'-(1-methylethyl)-1,3,5-triazine-2,4-diamine							
Bentazone	Herbicide	3-isopropyl-1H-2,1,3-benzonthiadiazain-(4)3H-one 2,2-dioxide							
Carbofuran	Insecticide	2,3-dihydro-2,2-dimethyl-7-benzofuranyl methylcarbamate							
Chlorimuron	Herbicide	Ethyl-2-[[[[4-chloro-6-methoxy-2-(pyrimidinyl)amino]carbonyl]amino]sulfonyl]							
		benzoic acid							
Clomazone	Herbicide	2-[(2-chlorophenyl)methyl]-4,4-dimethyl-3-isoxazolidinone							
Clopyralid	Herbicide	3,6-dichloropicolinic acid							
Chlorpyrifos	Insecticide	O, O-diethyl-O-(3,5,6-trichloro-2-pyridyl) phosphorothioate							
Cyanazine	Herbicide	[2-4-chloro-6-(ethylamino) 1,3,5-triazin-2-yl]amino-2-methylpropanenitrile							
Diazinon	Insecticide	O, O-dimethyl O-2-isopropyl-6-methylpyrimidin4-yl phosphorothioate							
Dicamba	Herbicide	3,6-dichloro-2-methoxybenzoic acid							
Diclofop	Herbicide	$(\pm)$ -2-(4-(2,4-dichlorophenoxy)phenoxy)propanoic acid							
Diclosulam	Herbicide	N-(2,6-dichlorophenyl)-5-ethoxy-7-fluoro-(1,2,4)triazolo(1,5-c)pyrimidine-2-							
		sulfonamide							
Diketonitrile	Herbicide	2-cyclopropyl-3-(2-mesyl-4-trifluoromethylphenyl)-3-oxopropanenitrile							
Dimethipin	Defoliant	2,3-dihydro-5,6-dimethyl-1,4-dithiin 1,1,4,4-tetraoxide							
Fluometuron	Herbicide	N,N-dimethyl-N'-[3-(trifluoromethyl)-phenyl]-urea							
Fonofos	Insecticide	O-ethyl S-phenyl ethylphosphonodithioate							
Glyphosate	Herbicide	N-(phosphonomethyl)glycine							
Imazapic	Herbicide	(RS)-2-(4,5-dihydro-4-isopropyl-4-methyl-5-oxoimidazol-2-yl)-5-methylnicotinic acid							
Imazapyr	Herbicide	2-(4,5-dihydro-4-methyl-4-(1-methylethyl)-5-oxo-1 <i>H</i> -imidazol-2-yl)-3-pyridine carboxylic acid							
Imazaquin	Herbicide	2-[4,5-dihydro-4-methyl-4-(1-methylethyl)-5-oxo-1 <i>H</i> -imidazol-2-yl]-3-							
Imazethapyr	Herbicide	$(\pm)$ -2-[4,5-dihydro-4-methyl-4-(1-methylethyl)-5-oxo-1 <i>H</i> -imidiazol-2-yl]-5-ethyl-							
Isoproturon	Harbiaida	3 (4 isopropylphonyl) 1 1 dimethyluros							
Isoproturole	Herbicide	5 cyclopropyl 1.2 isovazal 4 yl alpha alpha alpha triffuoro 2 mesyl p tolyl ketope							
Linuron	Herbicide	3-(3.4-dichloronbenyl)-1-methoxy-1-methylurea							
MCPA	Herbicide	4 chloro 2 methyphenoxy acetic acid							
Metalachlor	Herbicide	$\frac{1}{2}$ chloro N (2 ethyl 6 methylphenyl) N (2 methoxy 1 methylethyl)acetamide							
Metamitron	Herbicide	$4_4$ -amino-3-methyl-6-nhenyl-1.2.4-triazin-5( $4H$ )-one							
Metribuzin	Herbicide	4-amino 5 methyl 6 phenyl 1,2,4 thazh 5(411) one 4-amino 6-(1 1-dimethylethyl) 3-(methylthio) 1.2 4-triazine 5(4H)-one							
Norflurazon	Herbicide	4-chloro-5(methylamino)-2-(3-(trifluoromethyl)nhenyl)-3(2H)-pyridazinone							
Oxyfluorfen	Herbicide	2-chloro-1-(3-ethoxy-4-nitrophenoxy)-4-(trifluoro-methyl) benzene							
Pendimethalin	Herbicide	N-(1-ethylpropyl)-3 4-dimethyl-2 6-dinitrobenzenamine							
Propachlor	Herbicide	2-chloro- <i>N</i> -isopropylacetanilide							
Simazine	Herbicide	2-chloro-4 6-bis[ethylamino]-s-triazine							
Sulfentrazone	Herbicide	N-[2 4-dichloro-5-[4-(difluoromethyl)-4 5-dihydro-3-methyl-5-oxo-1H-1 2 4-							
		triazol-1-yl]phenyl]methanesulfonamide							
Terbutos	Insecticide	S-(((1,1-dimethylethyl)thio)methyl) $O$ , $O$ -diethyl phosphorodithioate							
Terbuthylazine	Herbicide	N-2-tert-butyl-6-chloro- $N$ -4-ethyl-1,3,5-triazine-2,4-diamine							
Thidiazuron	Detoliant	<i>N</i> -phenyl- <i>N</i> -1,2,3-thidiazol-5-yl-urea							
Tribufos	Defoliant	S,S,S-tributyl phosphorotrithioate							
2,4 D	Herbicide	(2,4-dichlorophenoxy)acetic acid							
2,4-Dichlorophenol	Herbicide	2,4-Dichlorophenol							
	precursor								
			Soil p	roperties					
-------------	---------------	-------	--------	------------------------	------	-------------	-------------------	--------------	----------------------
Destisite	<b>T</b> :111	Depth	Class	Organic carbon (OC)	-11	$K_{\rm D}$	$K_{\rm F}$	$K_{\rm OC}$	Deferrer
Pesticide	Thage		Clay	g Kg	рп		L · mg · · · · kg	L Kg OC	(Farri at al
Acetochlor	NT	0–10	-	16.6	-	2.7	-	166	(Ferri et al., 2002)
	CT		-	13.3	-	1.7	-	126	2002)
Acifluorfen	NT	0–10	-	10.2	5.6*	-	0.8	_	(Gaston and
	NT	10-10	_	5.6	5.3*	-	0.5	_	Locke, 2000)
	NT	20-30	_	4.4	5.7*	-	0.6	_	
	CT	0-10	-	8.7	5.8*	-	0.8	_	
	CT	10-20	_	6.4	5.8*	-	0.5	_	
	CT	20-30	-	4.9	5.8*	-	0.5	-	
Alachlor	NT	0–5	290	16.7	5.5*	5.4	5.6	323	(Locke, 1992)
	CT	0–5	260	10.2	5.5*	3.5	3.6	342	
Alachlor	NT	0–5	_	72.5	4.7*	_	6.0	-	(Clay et al.,
	NT	5-10	-	29.0	5.0*	-	3.6	_	1991)
	NT	10-15	_	29.5	6.2*	-	4.0	_	
	NT	15-20	_	32.5	6.4*	-	3.7	_	
	CT	0–5	_	30.0	5.5*	-	3.6	_	
	CT	5-10	_	26.0	6.0*	-	4.0	_	
	CT	10-15	_	29.0	5.2*	-	3.9	_	
	CT	15-20	-	28.0	5.7*	_	4.0	-	
Alachlore	$NT + B^2$	0-15	_	8.4	_	0.9	_	110	(Xue et al.,
	$NT + V^2$		_	8.6	_	1.1	_	130	1997)
	NT		_	6.7	-	1.0	_	143	
	CT + B		_	4.7	-	0.5	_	99	
	CT + V		_	5.7	_	0.5	_	95	
	CT		-	6.3	-	0.6	-	91	
Atrazine	NT	0–2.5	_	19.1	5.2	7.8	_	_	(Ghadiri et al.,
	NT	2.5-5	-	15.1	5.0	7.1	-	_	1984)
	NT	5-10	-	10.4	5.8	7.0	-	_	
	NT	10-15	-	10.4	6.2	7.3	-	_	
	NT	15-20	-	10.4	6.3	7.0	-	_	
	NT	20-25	-	8.1	6.5	6.2	-	_	
	NT	25-30	-	7.0	6.7	6.1	-	_	
	NT	30-35	-	6.4	6.8	6.1	-	_	
	NT	35-40	-	5.8	6.9	5.8	-	-	
	CT	0-2.5	-	12.8	5.6	7.4	_	_	
	CT	2.5-5	-	12.8	5.2	7.5	_	_	
	CT	5-10	-	11.6	5.7	7.6	-	_	
	CT	10-15	_	11.0	6.2	6.7	-	_	
	CT	15-20	-	10.4	6.3	6.5	_	-	
	CT	20-25	-	8.1	6.6	6.5	_	-	
	CT	25-30	-	7.0	6.7	6.0	_	-	
	CT	30-35	-	5.2	6.9	5.7	_	-	
	CT	35-40	_	4.7	7.0	5.5	_	_	

 Table 3
 Sorption properties of pesticides under different tillage practices

#### Table 3 (continued)

			Soil p	roperties					
				Organic					
		Depth		carbon (OC)		KD	K <sub>F</sub>	KOC	
Pesticide	Tillage <sup>1</sup>	cm	Clay	g kg <sup>-1</sup>	pН	$L kg^{-1}$	$L^{n_{\rm F}} mg^{(1-n_{\rm F})} kg^{-1}$	L kg <sup>-1</sup> OC	Reference
Atrazine	NT	0-15	_	10.6	_	1.7	-	_	(Novak et al.,
	CT		_	9.8	_	1.4	-	_	1996)
Bentazon	NT	0-10	_	10.2	5.6*	0.03	0.04	_	(Gaston et al
Dentazon	NT	10_20	_	56	5.3*	0.05	0.04	_	(Gaston et al., 1996)
	NT	20 30		1.0	5.7*				1990)
	CT	20-30	_	4.4 8 7	5.8*			_	
	СТ	10 20	_	6.1	5.0*			_	
	СТ	20 20	_	0.4	5.0*			_	
	CI	20-30	_	4.9	5.8			—	
Chlorimuron	NT	0–7.5	280	30.6	6.6*	-	1.0	31	(Reddy et al.,
	CT		250	17.8	6.4*	-	0.6	31	1995a)
	NT		330	46.0	6.2*	-	6.5	142	
	CT		260	19.2	4.5*	-	2.1	108	
	NT		-	15.3	5.4*	_	2.1	141	
	CT		_	13.1	5.7*	-	1.6	125	
Clonyralid	NT $(0)^{3}$	0-5	145	23.7	5 1	1.0	_	21	(Shang and
Clopyrand	CT	0–5	145	23.7	5.8	0.3	_	21	Arshad.
	NT (8)		532	47.0	49	1.6	_	51	1998)
	CT		520	44.5	5.4	2.1	_	32	
		o -				2.5			
Cyanazine	NT (10)	0–5	230	21.4	5.2	3.5	-	165	(Reddy et al.,
	CI		210	16.0	5.3	2.2	-	140	1997b)
Dicamba	NT (9)	0–5	145	23.7	5.1	0.2	-	13	(Shang and
	CT		145	22.7	5.8	Undetected	_	Undetected	Arshad,
	NT (8)		532	47.0	4.9	1.4	-	23	1998)
	CT		520	44.5	5.4	1.6	-	23	
Diclosulam	NT	0-10	355	19.9	5.0*	1.9	1.8	98	(Lavorenti
	СТ	0-10	341	12.8	4.4*	2.0	1.9	159	et al., 2003)
	NTT 1 A 1 2	0.2		12.0	5.0		2.0		
Fluometuron	$NI + Adv^2$	0-3	-	13.9	5.2 5.5	-	2.0	_	(Gaston et al.,
	NT + Auv	5-0 0 2	_	0.0 15.6	5.5	_	1.0	_	2001)
	NT + V	0-3 3-6	_	73	5.5	_	1.0	_	
	NT+B	0_3	_	11.2	6.1	_	1.2	_	
	NT+B	3-6	_	47	5.5	_	1.3	_	
	CT+Adv	0-3	_	7.4	5.7	_	0.8	_	
	CT+Adv	3–6	_	7.0	5.2	_	1.2	_	
	CT+V	0–3	_	7.4	5.5	_	0.9	_	
	CT+V	3–6	_	6.6	5.0	_	1.1	_	
	CT+B	0–3	-	7.8	5.8	-	1.0	_	
	CT+B	3–6	-	5.8	5.1	-	0.8	_	
Fluometuron	NT	0-4	150	20.0	51	22	_	_	(Brown et al
Tuometuron	NT	4-8	150	10.0	5.5	1.7	_	_	(DIOWI Ct al., 1994)
	NT	8-15	160	9.0	6.5	1.4	_	_	
	СТ	0–4	130	13.0	5.5	1.7	_	_	
	CT	4-8	140	13.0	5.7	2.1	_	_	
	CT	8-15	150	11.0	6.3	1.4	-	-	
	NT+V	0–4	150	25.0	4.7	3.0	-	_	
	NT+V	4-8	150	11.0	4.9	1.0	-	_	
	NT+V	8-15	150	9.0	5.7	1.3	-	_	
	CT+V	0–4	140	15.0	5.0	2.1	-	_	
	CT+V	4-8	150	15.0	5.0	2.2	-	-	
	CT+V	8–15	150	11.0	5.6	1.6	-	_	

#### Table 3 (continued)

			Soil pr	operties					
Pesticide	Tillage <sup>1</sup>	Depth cm	Clay	Organic carbon (OC) $g kg^{-1}$	pН	$K_{\rm D}$ L kg <sup>-1</sup>	$K_{ m F}$ ${ m L}^{n_{ m F}}{ m mg}^{(1-n_{ m F})}{ m kg}^{-1}$	$K_{\rm OC}$ L kg <sup>-1</sup> OC	Reference
		0.0	2	0.0	1	0		0	(Zablotowicz
Fluometuron	NT (11)	0-2	-	30.1	-	-	7.2	-	et al
	NT	2-5	-	11.8	-	-	2.0	-	2000)
	NT	5-10	-	5.9	-	-	1.4	-	2000)
	NT	10-25	-	3.3	-	-	1.5	-	
	CT	0-2	-	20.3	-	-	1.9	—	
	CT	2-5	-	11.2	-	-	2.2	-	
	CT	5-10	-	6.9	-	-	1.0	-	
	CT	10–25	-	4.4	-	-	1.2	-	
Fluometuron	NT	0-8	130	24.5	5.1*	2.0	_	80	(Suba and
	CT	0–8	100	14.2	5.2*	1.3	_	92	Essington, 1999)
Fluometuron	NT	0–2	_	13.9	6.7	2.4	_	_	(Locke et al.,
	NT	2-10	_	6.1	5.9	1.6	_	_	2005)
	NT+Rg <sup>2</sup>	0–2	_	19.8	6.0	5.0	-	_	
	NT+Rg	2-10	_	6.9	5.7	1.4	-	_	
	CT	0-2	_	8.0	6.5	1.7	_	_	
	CT	2-10	_	6.0	5.8	1.6	-	_	
	CT+Rg	0-2	_	9.5	6.4	2.1	_	_	
	CT+Rg	2-10	_	7.3	5.7	2.0	_	_	
	RT	0-2	_	12.2	6.5	0.6	_	_	
	RT	2-10	_	6.7	6.7	0.4	_	_	
	RT+S	0-2	_	12.6	6.6	0.7	_	_	
	RT+S	2-10	_	7.1	6.6	0.5	_	_	
	CT	0–2	_	6.4	6.1	0.4	_	_	
	СТ	2-10	_	8.4	6.4	0.4	_	_	
	CT+S	0-2	_	6.8	6.1	0.4	_	_	
	CT+S	2-10	_	7.9	6.6	0.5	_	_	
	RT	0-2	_	10.3	5.5	0.5	_	_	
	RT	2-10	_	9.5	5.5	0.5	_	_	
	RT+S	0-2	_	8.8	5.5	0.4	_	_	
	RT+S	2-10	_	8.5	5.9	0.5	_	_	
	CT	0-2	_	9.0	5.6	0.4	_	_	
	СТ	2-10	_	8.6	5.4	0.4	_	_	
	CT+S	0-2	_	8.7	5.7	0.4	_	_	
	CT+S	2-10	-	8.6	5.5	0.4	_	_	
MCPA	NT (9)	0–5	145	23.7	5.1	2.0	_	63	(Shang and
	CT		145	22.7	5.8	0.7	-	40	Arshad,
	NT (8)		532	47.0	4.9	7.3	-	168	1998)
	CT		520	44.5	5.4	10.4	-	182	
Metolachlor	NT	0-5	_	15.4	_	_	3.5	230	(Ding et al.
	NT	10-15	_	3.5	_	_	1.3	360	2002b)
	СТ	0-5	_	8.5	_	_	2.0	240	20020)
	CT	10–15	-	5.4	-	-	1.4	256	
Norflurazon	NT	0-8	130	24.5	5.1*	12.1	_	496	(Suba and
- ormanuzon	СТ	0-8	100	14.2	5.2*	6.5	-	456	Essington,

#### Table 3 (continued)

			Soil pr	operties					
Pesticide	Tillage <sup>1</sup>	Depth cm	Clay	Organic carbon (OC) g kg <sup>-1</sup>	рН	$K_{\rm D}$ L kg <sup>-1</sup>	$\frac{K_{\rm F}}{L^{n_{\rm F}} {\rm mg}^{(1-n_{\rm F})}  {\rm kg}^{-1}}$	$K_{\rm OC}$ L kg <sup>-1</sup> OC	Reference
Norflurazon	NT	0–2	-	13.9	6.7	2.2	-	_	(Locke et al.,
	NT	2-10	-	6.1	5.9	1.6	-	-	2005)
	NT+Rg	0-2	-	19.8	6.0	3.6	-	-	
	NT+Rg	2-10	-	6.9	5.7	1.6	-	-	
	CT	0-2	-	8.0	6.5	1.6	-	-	
	CT	2-10	-	6.0	5.8	1.8	-	-	
	CT+Rg	0-2	-	9.5	6.4	2.3	-	-	
	CT+Rg	2-10	-	7.3	5.7	2.1	-	-	
Sulfentrazon	NT	0-7.5	280	30.6	6.6*	0.9	1.2	30	(Reddy and
	CT		250	17.8	6.4*	0.8	1.0	47	Locke, 1998
	NT		330	46.0	6.2*	3.2	3.4	71	
	CT		260	19.2	4.5*	2.9	2.6	153	
	NT		230	21.4	5.2*	2.3	3.3	96	
	CT		210	16.0	5.3*	1.5	1.8	106	
2,4-	NT (9)	0-3	188	27.0	6.6*	4.2	_	155	(Düring, 2002)
Dichlorophenol	NT	3-10	177	16.7	6.6*	2.2	-	132	
	NT	10-25	168	10.3	6.4*	1.3	-	122	
	CT	0-3	132	11.3	6.1*	1.7	-	150	
	CT	3-10	155	11.6	6.1*	1.6	-	140	
	CT	10-25	143	11.3	6.2*	1.4	-	126	

<sup>1</sup> Codes of tillage practices are reported in Table 1.

<sup>2</sup> Adv: weed residues; B: wheat residues; Rg: ray-grass residues; S: rye residues; V: vetch residues.

<sup>3</sup> Numbers in parenthesis indicate the age in years of the tillage system.

\* pH CaCl<sub>2</sub>.

by the next 30 mm of water. At the end of their experiment, most of the applied chemicals had been washed off from the mulch. Timing between treatment and first rainfall also strongly influences the quantity of pesticides that could be washed from the mulch. It was found to be the highest after heavy rainfall occurring within two weeks following the treatment (Sadeghi and Isensee, 1996; 1997). Depending on rainfall quantity and intensity, between 70 and 96% of intercepted atrazine were released between the 1st and 3rd weeks after treatment (Ghadiri et al., 1984; Isensee and Sadeghi, 1994). However, this wash-off directly depends on the retention capacities of the pesticide on crop residues (Gaston et al., 2001). For example, metolachlor, once intercepted, seems to form stronger and less reversible bonds with crop residues than acetochlor or alachlor (Banks and Robinson, 1986). In some cases, a gradual release of the pesticide from the residues by wash-off may lead to an increase in its efficacy (Dao, 1991), but could also increase crop injuries (Barnes et al., 1989; Loux et al., 1989; Mills and Witt, 1991).

#### 2.2 Organic Carbon Content Effect

One of the main changes related to the implementation of conservation tillage compared with conventional tillage is the redistribution of organic carbon in the soil (Balesdent et al., 1990; Tebrügge and During, 1999). Generally, organic carbon content increases in surface soil due to the presence and decomposition of the mulch and gradually decreases with depth (Lal et al., 1994; Pinheiro et al., 2004; Six et al., 1999). For most pesticides, organic carbon content and adsorption are positively correlated, resulting in a higher adsorption in surface soil under conservation tillage than under conventional tillage (Table 3). For molecules with low sorption capacity, such as bentazon (Gaston et al., 1996) or diclosulam (Lavorenti et al., 2003), the effect of tillage may not be significant. Very few studies aimed at analysing the effects of tillage on pesticide sorption kinetics. For acifluorfen (Gaston and Locke, 2000), chlorimuron (Reddy et al., 1995a) and cyanazine (Reddy et al., 1997b), no effect of tillage system was found on sorption kinetics and most of the

sorption occurred during the first hour of contact. For some molecules, such as alachlor (Locke, 1992) and sulfentrazone (Reddy and Locke, 1998), sorption was faster under conservation tillage and was positively correlated with higher soil organic matter contents.

Tillage management also modifies pesticide desorption. For sulfentrazone and alachlor, whatever the tillage system, desorption was found to be hysteretic, but a greater proportion of the molecule, representing approximately 35% of the applied sulfentrazone, remained sorbed under no-tillage (against 20% under conventional tillage) (Locke, 1992; Reddy and Locke, 1998). For cyanazine (Reddy et al., 1997b), desorption was also more reversible under conventional tillage but, when ryegrass residues were added to the soil samples, desorption was similar under conventional and conservation tillage. As indicated by Locke (1992), desorption tends to decline with increasing contact time between soil and pesticide due to diffusion processes within the soil.

However, considering only the quantitative increase in organic carbon content in surface soil under conservation tillage is not sufficient to explain the increase in pesticide sorption found in some studies (Ding et al., 2002b; Novak et al., 1996; Zablotowicz et al., 2000). Although the analytical procedures used to extract soil organic matter can induce artefacts by modifying its molecular structure (Ding et al., 2002b; Salloum et al., 2001), it appears that the nature of soil organic matter and, hence, its reactivity, are also affected by tillage (Ding et al., 2002a; Sleutel et al., 2007; Tatzber et al., 2008) and by the nature of crop residues (Ding et al., 2006). In soil organic matter collected under a no-tillage system, Preston et al. (1994) found a higher number of reactive functional groups compared with that collected under a conventional tillage system. Ding et al. (2002a) found that humic acids and humin from conservation tillage contained more aliphatic carbon and less aromatic carbon than under conventional tillage. A larger amount of amino acids were also linked to these humic acids under no-tillage, whereas they were more linked to fulvic acids under conventional tillage (Szajdak et al., 2003). Moreover, reactive/recalcitrant peak ratios indicated that humic acids in the topsoil (0-5 cm-depth) were more biologically active under conservation tillage than under conventional tillage. With depth, the soil organic matter composition changes. Humification processes are more advanced in deeper soil layers and lead to

an increase in aromaticity with the development of polycondensed rings (Ding et al., 2002a). The reactivity of organic matter fractions may also affect pesticide desorption. As an example, desorption of metolachlor from conventional tillage humic acids was less significant than from conservation tillage humic acids (Ding et al., 2002b). The hysteresis index (which is the ratio of Freundlich exponents for desorption and sorption) was lower for humic acids under conventional tillage, indicating that sorbed metolachlor molecules were more difficult to desorb. This may be caused by higher aromatic carbon contents in conventional tillage humic acids (Ding et al., 2002b; Xing, 2001).

The nature of crop residues also influences the composition of soil organic matter. For example, rye residues tend to form more aromatic and less aliphatic humic acids than a mix of vetch/rye residues (Ding et al., 2006), which could potentially affect pesticide behaviour and efficacy in soil (Ding et al., 2002a; Nanny and Maza, 2001). However, to our current knowledge, little work has been published on the effects of these changes in soil organic matter composition due to tillage on pesticide sorption. Ding et al., (2002b) highlighted a stronger sorption of metolachlor by humin than by humic acids, and humin content was found to be higher under no-tillage systems (Stearman et al., 1989). For humin, which is a highly condensed form of soil organic matter, several studies showed that sorption was more variable and that sorption isotherms were less linear (Xing and Pignatello, 1997; Yuan and Xing, 1999). In contrast, Stearman et al. (1989) observed a higher sorption of metribuzin and oxyfluorfen on fulvic and humic acids, which are abundant in conventional tillage, than on humin, which is abundant in conservation tillage. Dissolved organic carbon content is most often higher under conservation tillage and is mainly composed of small molecules, with a size similar to that of fulvic acids (Suba and Essington, 1999). In some studies, sorption of pesticides on dissolved organic carbon was found to be an efficient way to increase the mobility of some pesticides in soils, e.g. atrazine and 2,4D, leading to a significant transport of these chemicals through the soil profile (Chin et al., 1990; Gao et al., 1998; Lafrance et al., 1994; Li et al., 2005). In another study, sorption of fluometuron and norflurazon on the dissolved organic carbon formed under conservation tillage did not lead to a higher leaching of these molecules (Suba and Essington, 1999).

#### 2.3 Soil pH Effect

Soil pH may be differently modified by tillage techniques: conservation tillage leads to an increase in pH (Reddy and Locke, 1998), no change (Comia et al., 1994; Rasmussen, 1991) or, more often, to a decrease in pH, especially in surface soil due to the accumulation of organic matter and fertilisers (Arshad et al., 1999; Doran, 1980; Levanon et al., 1994). For many pesticides, sorption is strongly influenced by soil pH and tends to increase when soil pH decreases (Barriuso and Calvet, 1992; Barriuso et al., 1992; Grey et al., 1997). In sandy soils, it has been found that, for similar organic carbon contents between tillage treatments, acidification in conservation techniques could significantly increase the sorption of several molecules according to their pKa (Shang and Arshad, 1998). For s-triazines (weak bases), the decrease in pH causes their protonation, promoting sorption on organic matter (Senesi and Testini, 1982) and decreasing their herbicide activity.

## 2.4 Other Effects

Tillage techniques modify other soil properties (Fig. 4) whose effects on retention mechanisms are still poorly understood. Due to the presence of a mulch, water content in conservation techniques is generally higher than in conventional techniques (Drury et al., 1999; Shelton et al., 1998). This increase in soil water content led, in some cases, to a decrease in pesticide retention (Dao and Lavy, 1991; Walker, 1971). In other cases, it led to an increase in adsorption that could be explained by a decrease in the hydrophobicity of the organic compounds and an access to sorption sites facilitated by the presence of water (Berglof et al., 2000; Ochsner et al., 2006). Soil temperature is also modified by the mulch under conservation tillage. The mulch intercepts light energy, thus reducing temperature at the soil surface in conservation techniques, from 1 to 5°C (Bragagnolo and Mielniczuk, 1990; Grant et al., 1990; Gupta et al., 1988). However, to our current knowledge, there is no published data about the consequences of soil water content and temperature modifications by tillage management on pesticide retention.

#### 2.5 Sorption of Metabolites

Little information is available on tillage effects on metabolite retention. According to their molecular composition, metabolites can have lower sorption capacities, e.g. cyanazine metabolites (Reddy et al., 1997a), or higher sorption capacities, e.g. atrazine and bentazon metabolites (Clay and Koskinen, 1990; Gaston et al., 1996) than the parent compound. For cyanazine metabolites, as observed for the parent compound, sorption was higher under conservation tillage than under conventional tillage.

# 2.6 Summary and Recommendations

Regarding pesticide retention processes, most studies highlighted the role of the mulch in conservation tillage systems which intercepted the molecules. In a general way, tillage systems act on pesticide retention mainly through their effect on the distribution of organic matter in soil. Recent works on the characterisation of organic matter fractions and their reactivity towards pesticides show significant influences of the type of tillage and nature of crop residues. Using recent analytical technologies, these studies on the characterisation of organic compounds found in soils or crop residues are a significant advance in the qualitative evaluation of the effects of agricultural practices on pesticide retention. To progress in this direction, further research should not be focused on tillage management only, but rather on the whole cropping systems. As a matter of fact, other management options such as crop rotation, cover crops and manure application play important roles in the dynamics of soil organic matter. Regarding conservation tillage systems, research efforts should be made to clarify the indirect effects of the mulch via modifications of pH, water and temperature dynamics in the underlying soil.

# **3 Degradation**

Understanding pesticide degradation in soils is a key step for assessing their persistence and their risks of transfer in the environment. Degradation studies can be carried out in the field, thus integrating a set of non-controlled phenomena such as fluctuations in temperature, soil water content or radiation (Table 4), or in the laboratory, where soil samples are kept under controlled conditions (Table 5). In all cases, degradation is conditioned by a combination of factors, i.e. microflora, organic matter, water content, temperature and pH, which are directly influenced by tillage (Fig. 4). Biotic degradation of pesticides is most often seen as quantitatively more important than abiotic degradation. In conservation tillage, the total microbial biomass is generally larger than in conventional tillage (Biederbeck et al., 1997; Dalal et al., 1991; Doran, 1980), but it remains strongly dependent on soil conditions. Fungi populations, which were found to be efficient in pesticide degradation (Barr and Aust, 1994), are generally larger in conservation tillage and their biomass increases as degree of tillage is reduced (Drijber et al., 2000; Kabir, 2005). These biological differences due to tillage have effects, sometimes contradictory, on the degradation of pesticides in the mulch and in the soil.

# 3.1 Pesticide Degradation in the Mulch

The interception by the mulch in conservation techniques can modify the persistence of the applied pesticides. These effects are sometimes negligible (Banks and Robinson, 1982), but more often they significantly affect the fate of pesticides in soil. Crop residues may have higher microbial activity and degradation capacity than the soil, thus reducing the concentrations of pesticide that reach the soil surface (Locke et al., 2005; Zablotowicz et al., 1998). Moreover, depending on the nature of the pesticide, interception by the mulch may generate photodegradation, thus reducing the persistence of the molecules (Selim et al., 2003). In other works, crop residues increased the residence time of pesticides because of the competition between retention and degradation processes, and a gradual release of the molecules by desorption was observed (Mazzoncini et al., 1998). By acting as a physical protector of the soil surface, crop residues may also limit the air flow between the soil and the atmosphere and within the soil, thus slowing down the activity of degrading microorganisms (Sorenson et al., 1991).

Depending on the nature of crop residues, the degradation of molecules can be affected by the presence of a mulch, but in contrasted ways. For example, in no-tillage, vetch residues accelerated the degradation of metolachlor by from 1.5 to 3 times, but had no effect on the degradation of atrazine (Teasdale et al., 2003). On the contrary, according to laboratory studies, vetch residues seemed to slow down fluometuron degradation compared with soil samples without vetch residues (Brown et al., 1994; Zablotowicz et al., 1998) or other types of residues such as wheat residues (Gaston et al., 2001), rye residues (Zablotowicz et al., 1998), or ray grass residues (Locke et al., 1995). This slowdown could be due to the abundance of nitrogen in legumes (Fabaceae). This nitrogen would be preferentially used by the microorganisms to the detriment of the N sources contained in pesticides. In the same way, Gan et al. (1996) observed a lower degradation rate of atrazine after an addition of nitrogen. However, experiments on undisturbed soil columns with simulated rainfall, that allowed nitrogen leaching and thus reduced the amount bioavailable in the soil column, relativised these results and no negative effect of vetch residues on fluometuron degradation was found (Gaston et al., 2003). This experiment provided an explanation of the difference between the results obtained in laboratory incubations where leaching of nitrogen is not possible (Gaston et al., 2001) and those obtained in the field where N can leach out through the soil (Brown et al., 1996).

## 3.2 Pesticide Degradation in Soil

Tillage practices were found to modify significantly pesticide degradation in soil, but in contrasted ways according to studies. Degradation of several pesticides was indeed found to be slower (Brown et al., 1994; Gaston and Locke, 2000; Otto et al., 1997; Ulbrich et al., 2005), equivalent (Gaynor et al., 1998; Locke et al., 1996; 2005; Monks and Banks, 1993; Reddy et al., 1995b; Renner et al., 1998), or faster (Gaston et al., 2001; 2003; Lavorenti et al., 2003; Levanon et al., 1994; Sadeghi and Isensee, 1997) under conservation tillage than under conventional tillage. Several reasons could explain these differences. To explain a lower degradation under conservation

Table 4	Field	studies	of pesticide	degradation	under	different	tillage practices	s

			Soil proper	ties			
		Depth		Organic carbor	1	$DT_{50}^{2}$	
Pesticide	Tillage <sup>1</sup>	cm	Clay	$g kg^{-1}$	pН	d	Reference
Alachlor	NT CT	0–30 0–30		-	_	<1 <1	(Weed et al., 1998)
Alachlor en- capsulated alachlor	NT CT NT CT	0-110 0-110 0-110 0-110	- - -		- - -	8-41 8-41 4-20 4-20	(Gish et al., 1994)
Atrazine	NT CT	0–10	_	_		50 42	(Ghadiri et al., 1984)
Atrazine	NT CT	0–50 0–70	_	_	_	71 73	(Gish et al., 1991)
Atrazine en- capsulated atrazine	NT CT NT CT	0-110 0-110 0-110 0-110		- - -	- - -	36 36 110 110	(Gish et al., 1994)
Atrazine	NT NT+Rg <sup>3</sup> CT CT+Rg	0–10 0–10 0–10 0–10	390	14.5	5.4	45, 36, 56 <sup>4</sup> 40, 33, 54 46, 34, 54 35, 59, 24	(Gaynor et al., 2000)
Atrazine	NT RT top RT furrow CT	0–10	- - -	- - -	- - -	43, 56, 43, 35 33, 75, 53, 36 31, 47, 53, 35 33, 62, 58, 47	(Gaynor et al., 1998)
Atrazine	NT CT	0–50	17–26 16–24	11.0 7.0	6–6.5 4.8–5	35, 25,12, 23 37, 21, 29, 18	(Isensee and Sadeghi, 1994)
Atrazine	NT CT	0–150	55–166 59–179	0.1–9.7 1.2–4.4	5.3–6.1 5.5–6.7	149 215	(Heatwole et al., 1997)
Atrazine	$NT (7)^{5}$ $CT$ $CT \rightarrow NT$ $NT$ $(7) \rightarrow CT$	0–50	- - -	- - -	- - -	12 22 -, 21, 32 -, 17, 23	(Sadeghi and Isensee, 1996)
Clomazon	RT CT	0–10	230	23.2	6.0	52, 91 58, 83	(Curran et al., 1992)
Chlorimuron	NT CT NT CT		260 260 220 220	10.4 10.4 8.7 8.7	7.8 7.8 6.3 6.3	32, 14, 22 44, 13, 18 17, 82 14, 22	(Baughman et al., 1996)
Cyanazine	NT CT	-0-50 0-70	_	_		13 13	(Gish et al., 1991)
Fluometuron	NT, NT+Rg CT, CT+Rg	0–2	_	_		7–15	(Locke et al., 2005)
Fluometuron	NT NT+V <sup>2</sup> CT CT+V	0–8	150 150 130 140	8.7 10.4 7.6 8.7	5.2 4.8 5.6 5.0	30, 23 38, 19 24, 26 30, 25	(Brown et al., 1996)
Imazapic	NT (3) CT NT	0–10 0–10 0–10	780 280	20.3 27.3	4.7 5.8	66 45 35	(Ulbrich et al., 2005)
Imazapyr	CT NT (3) CT NT	0–10 0–10 0–10 0–10	780	20.3	4.7	32 53 50 43	(Ulbrich et al., 2005)
Imazaquin	CT RT CT	0–10 0–10	280 230	27.3	5.8 6.0	45 39, 144 41, 117	(Curran et al., 1992)

#### Table 4 (continued)

			Soil proper	ties			
		Depth		Organic carbon	1	$-DT_{50}^{2}$	
Pesticide	Tillage <sup>1</sup>	cm	Clay	$g kg^{-1}$	pН	d	Reference
Imazaquin	NT CT	0–20	-	16.9, 19.8	5.9, 6.4	27, 33 53, 22	(Mills and Witt, 1991)
Imazethapyr	RT CT	0–10	230	23.2	6.0	82, 53 122, 56	(Curran et al., 1992)
Imazethapyr	NT CT	0–20	-	16.9, 19.8	5.9, 6.4	36, 40 40, 12	(Mills and Witt, 1991)
Isoproturon	NT RT CT	0–30 0–30 0–30	170 170 170	6.1 5.8 4.5		12 8 15	(Otto et al., 1997)
Metolachlor	NT RT CT	0–30 0–30 0–30	170 170 170	6.1 5.8 4.5		9 26 29	(Otto et al., 1997)
Metolachlor	NT CT	0–150	55–166 59–179	0.1–9.7 1.2–4.4	5.3–6.1 5.5–6.7	45 34	(Heatwole et al., 1997)
Metolachlor	NT RT top RT furrow CT	0–10	- - -		- - -	40, 65, 37, 30 23, 87, 41, 35 28, 42, 40, 28 32, 68, 43, 40	(Gaynor et al., 1998)
Metolachlor	NT NT+Rg CT CT+Rg	0-10 0-10 0-10 0-10	390	14.5	5.4	46, 42, 72 42, 40, 69 42, 45, 79 44, 44, 97	(Gaynor et al., 2000)
Metribuzin	NT NT+Rg CT CT+Rg	0-10 0-10 0-10 0-10	390	14.5	5.4	24, 27, 34 24, 26, 36 23, 29, 32 24, 29, 37	(Gaynor et al., 2000)
Metribuzin	NT CT NT	0–5 0–5 0–5		13.3 18.0	5.7 5.1	12, 11 5, 17 15, 15	(Sorenson et al., 1991)
	CT	0–5	-			13, 11	
Terbuthylazine	NT RT CT	0–30 0–30 0–30	170 170 170	6.1 5.8 4.5		25 32 22	(Otto et al., 1997)

<sup>1</sup> Codes of tillage practices are reported in Table 1.

<sup>2</sup> DT<sub>50</sub>: pesticide half-life.

<sup>3</sup> Rg: ray-grass residues; V: vetch residues.

<sup>4</sup> Commas are used to separate different years of study.

<sup>5</sup> Numbers in parenthesis indicate the age in years of the tillage system.

tillage, the most commonly mentioned phenomenon is the competition between retention and degradation. In conservation tillage, pesticide adsorption to the soil solid phase is generally increased and may lead to a decrease in the availability of the molecules for biological degradation (Zablotowicz et al., 2000). In some cases, microbiological activity in soil can be affected by a lower temperature (Sorenson et al., 1991) or higher soil acidity (Brown et al., 1994) under conservation tillage. Both of these consequences were found to increase sorption of pesticides and thus to reduce their bioavailability. For some authors, repeated fertiliser inputs (Gaynor et al., 1998; Ghadiri et al., 1984), the use of legumes as cover crops (Brown et al., 1994) and the absence of lime could lead to soil acidification and thus mask the effects of tillage practices, particularly for molecules of the s-triazine family whose sorption is highly sensitive to acidity. Other pesticides such as sulfonyl-ureas have their chemical stability reduced by acidification, leading to a faster degradation of these compounds under conservation techniques (Chapman and Cole, 1982). Global soil microbial activity can be correlated with mineralisation of the molecules (Lavorenti et al., 2003), but

$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$				Soil prop	erties	b							
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$				1	Organic		Water		Incubation		$CO_2^2$	$NER^2$	
Actification         NT $0-10$ $ 102$ $5.6^*$ $0.35$ $2.5$ $4.9$ $108$ $5.1^*$ $0.35$ $2.5$ $4.9$ $108$ $5.1^*$ $0.35$ $2.9$ $108$ $5.1^*$ $0.35$ $2.9$ $108$ $5.1^*$ $0.35$ $2.9$ $106$ $5.1^*$ $0.35$ $2.5$ $4.9$ $108$ $5.5$ $7$ $7$ $7$ $2.9$ $5.1^*$ $0.35$ $5.3^*$ $0.35$ $5.3^*$ $0.35$ $5.3^*$ $7$ $7$ $7$ $7$ $7$ $7$ $7$ $7$ $7$ $7$ $7$ $2$ $2$ $2.5^*$ $5.3^*$ $0.30$ $2.5$ $5.4^*$ $6.5$ $7$ $7$ $2$ $2$ $7$ $7$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ $2$ <th< th=""><th>Pesticide</th><th>Tillage<sup>1</sup></th><th>Depth cm</th><th>Clay</th><th>carbon g kg<sup>-1</sup></th><th>Hq</th><th>content g g^{-1}</th><th>Temperature °C</th><th>duration d</th><th><math>{ m DT}_{50}^2</math> d</th><th>% applied dose</th><th>% applied dose</th><th>Reference</th></th<>	Pesticide	Tillage <sup>1</sup>	Depth cm	Clay	carbon g kg <sup>-1</sup>	Hq	content g g^{-1}	Temperature °C	duration d	${ m DT}_{50}^2$ d	% applied dose	% applied dose	Reference
Alachlor         NT $0.5$ $ 116$ $5.1^*$ $0.35$ $25$ $54$ $6.5$ $13$ Alachlor         NT $0-30$ $                                                                                -$	Acifluorfen	CT N T	$\begin{array}{c} 0-10\\ 20-30\\ 0-10\\ 20-30\end{array}$		10.2 4.4 8.7 4.9	5.6* 5.7* 5.8* 5.8*	0.35	25	49	108 165 74 169	v 4 v	9 5 3 3	(Gaston and Locke, 2000)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Alachlor	NT (7) <sup>3</sup> CT	0-5 0-5	1 1	22.0 11.6	5.3* 5.1*	0.35	25	54	6.5 6.5	13 7	54 43	(Locke et al., 1996)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Alachlor	NT NT	$0-30 \\ 0-30$	1 1	1 1	1 1	1 1	1 1	1 1	§ §	1 1	1 1	(Weed et al., 1998)
Bentazon (soil NT         0-10         -         -         -         -         -         -         -         17-23 <sup>4</sup> 0.1           columns)         NT         10-20         -         -         -         -         -         17-23 <sup>4</sup> 0.1           columns)         NT         10-20         -         -         -         -         -         69         0.1           CT         0-10         -         -         15.3         5.4*         -         -         23-35         0.1           Razon         NT (9)         -         15.3         5.4*         -         -         23-35         12         12           NT (16)         -         10-7.5         -         13.1         5.7*         48         7         12           NT (18)         0-7.5         -         13.1         5.7*         48         9         14           NT (18)         0-7.5         -         16.3*         0.33         25         48         9         14           NT (18)         0-7.5         5.6*         0.33         25         48         9         14           NT (4)         -         - <t< td=""><td>Bentazon</td><td>CT NT</td><td>0-10 20-30 0-10 20-30</td><td>       </td><td>10.2 4.4 8.7 4.9</td><td>5.6* 5.7* 5.8* 5.8*</td><td>0.30</td><td>25</td><td>22</td><td>50 87 39 77</td><td>0 0 0 0 0</td><td>15 8 9</td><td>(Gaston and Locke, 1996)</td></t<>	Bentazon	CT NT	0-10 20-30 0-10 20-30		10.2 4.4 8.7 4.9	5.6* 5.7* 5.8* 5.8*	0.30	25	22	50 87 39 77	0 0 0 0 0	15 8 9	(Gaston and Locke, 1996)
BentazonNT (9) $ 15.3$ $5.4^*$ $48$ $7$ $12$ CT $ 15.3$ $5.4^*$ $ 15.3$ $5.4^*$ $48$ $9$ $17$ NT(16) $  13.1$ $5.7^*$ $48$ $9$ $18$ NT(16) $  48.0$ $6.3^*$ $9$ $18$ NT(18) $0-7.5$ $ 18.9$ $4.5^*$ $9$ $18$ NT(18) $0-7.5$ $ 17.8$ $6.4^*$ $0.33$ $25$ $48$ $9$ $14$ NT(13) $ 17.8$ $6.4^*$ $0.33$ $25$ $48$ $9$ $14$ NT(4) $ 20.4$ $6.3^*$ $0.33$ $25$ $48$ $9$ $14$ NT(4) $ 17.7$ $5.0^*$ $6.4^*$ $0.33$ $25$ $48$ $9$ $14$ NT(4) $ 20.4$ $6.3^*$ $0.33$ $25$ $48$ $9$ $14$ NT(3) $ 17.7$ $5.0^*$ $5.6^*$ $22$ $39$ $39$ $37$ CT $ 280$ $30.6$ $6.6^*$ $                                            -$ <t< td=""><td>Bentazon (soil columns)</td><td>CT NT CT</td><td>0-10 10-20 0-10 10-20</td><td></td><td>1 1 1 1</td><td>1 1 1 1</td><td>1 1 1 1</td><td>1 1 1 1</td><td>1 1 1 1</td><td>17–23<sup>4</sup> 69 23–35</td><td>0.1 0.1</td><td>15–17 8–12</td><td>(Gaston and Locke, 1996)</td></t<>	Bentazon (soil columns)	CT NT CT	0-10 10-20 0-10 10-20		1 1 1 1	1 1 1 1	1 1 1 1	1 1 1 1	1 1 1 1	17–23 <sup>4</sup> 69 23–35	0.1 0.1	15–17 8–12	(Gaston and Locke, 1996)
CT 250 17.8 6.4* - 12 NT 0.75 330 46.0 6.2* 0.31 - 6.3 - 11	Chlorimuron	NT (9) CT NT(16) CT NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(16) NT(17) NT(16) NT(16) NT(16) NT(17) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(18) NT(	0-7.5	280 280	2.21 13.1 48.0 13.1 17.8 17.7 17.7 8.7 8.7 8.7 30.6	5.7* 6.3* 5.0* 5.6* 5.8* 6.6* 6.6*	0.33	5	4 4 4 4 8 4 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	- 6612	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	2 59 55 50 65 65 2 52 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	(waguer et al., 1996) (Reddy et al., 1995b)
CT 260 19.2 4.5* 0.01 - 00 - 114		CT NT CT	0-7.5	250 330 260	17.8 46.0 19.2	6.4* 6.2* 4.5*	0.31	1	63	1 1 1	12 11 14	18 24 15	×

Table 5 (continu	(pər											
			Soil pro	perties								
				Organic		Water		Incubation		$CO_2^2$	$NER^2$	
Pesticide	Tillage <sup>1</sup>	Depth cm	Clav	carbon g kg <sup>-1</sup>	Ha	content g g g -1	Temperature °C	duration d	$\mathrm{DT}_{50}^2$ d	% applied dose	% applied dose	Reference
	, TN		,	5 g 15.3	$5.4^{*}$	0			I	16	24	
	CT		I	13.1	5.7*				I	16	24	
Diclosulam	NT	0-10	355	19.9	5.0*	60	25	119	67	14	29	(Lavorenti et al., 2003)
	CT	0-10	341	12.8	4.4*	% WHC <sup>5</sup>			87	11	24	())) <b>,</b>
Fluomeuron	NT(11)	0–2	I	30.1	I				20	I	24	(Zablotowicz et al., 2000)
	NT	2-5	I	11.8	I				19	I	25	
	NT	5 - 10	I	5.9	I				48	I	13	
	NT	10-25	I	3.3	Ι	0.33	28	25	50	I	6	
	CT	0-2	I	20.3	T				6	I	43	
	CT	2-5	I	11.2	I				11	I	31	
	CT	5 - 10	I	6.9	I				25	Ι	23	
	CT	10–25	I	4.4	I				94	I	7	
Fluometuron	$NT + Adv^{6}$	0–3	I	7.2	5.2				9	I	I	(Gaston et al., 2001)
	$NT+V^{6}$	0-3	I	9.1	5.5				44	I	I	
	$NT+B^{6}$	0–3	I	9.2	6.1	$33 \mathrm{kPa}^7$	25	60	6	I	I	
	CT+Adv	0–3	I	3.4	5.7				44	I	I	
	CT+V	0–3	I	4.6	5.5				74	Ι	Ι	
	CT+B	0–3	I	4.9	5.8				23	I	I	
Fluometuron	NT+Adv	0-7.5	I	13.9	5.2				6	I	I	(Gaston et al., 2003)
	N+V	0-7.5	Ι	15.6	5.5				6	I	I	
	NT+B	0-7.5	I	11.2	6.1	PS	25	109	6	I	I	
	CT+Adv	0-7.5	I	7.4	5.7				19	I	I	
	CT+V	0-7.5	I	7.4	5.5				12	I	Ι	
	CI+B	c:/-0	I	1.8	0.0				У	I	I	

Fluometuron	NT(11)	0-4	150	20.0	5.1				57	I	I	(Brown et al.,
												1994)
	NT	48	150	10.0	5.5				62	I	I	
	NT	8-15	160	9.0	6.5		30	84	49	I	I	
	CT	0-4	130	13.0	5.5				49	I	I	
	CT	4-8	140	13.0	5.7				55	I	I	
	CT	8-15	150	11.0	6.3	0.25			52	I	I	
	NT + V(11)	0-4	150	25.0	4.7				78	I	I	
	NT+V	4-8	150	11.0	4.9				90	I	I	
	NT+V	8-15	150	9.0	5.7				83	I	I	
	CT+V	0-4	140	15.0	5.0				71	I	I	
	CT+V	4-8	150	15.0	5.0				68	I	I	
	CT+V	8-15	150	11.0	5.6				52	I	I	
Imazaquin	SS(3)	I	660	16.9	6.4	0.25-	30–25	9	12–16	I	I	(Seifert et al.,
	CT	I				06.0				I	Ι	20018)
Sulfentrazone	NT	0-7.5	230	21.4	I	0.30	I	77	I	2	27	(Reddy and
												LOCKE, 1998)
	CT		210	16.0	I		I		I	2	23	
Codes of tills	- one contract one -	l ai lotto	Toble 1									

 $^1$  Codes of tillage practices are reported in Table 1.  $^2$  DT\_{50}: pesticide half-life; CO\_2: mineralisation; NER: non-extractable residues.

<sup>3</sup> Numbers in parenthesis indicate the age in years of the tillage system. <sup>4</sup> Hyphens are used to indicate the range of variation of values. <sup>5</sup> Water content expressed in % of the water-holding capacity (WHC). <sup>6</sup> Adv: weed residues; B: wheat residues; V: vetch residues.

 $^7$  Water content expressed with the corresponding matric potential (kPa).  $\ast$  pH CaCl<sub>2</sub>.

not systematically (Reddy and Locke, 1998). The increase in soil microbial activity under conservation techniques did not always mean that specific microbial populations involved in the degradation of a molecule were more abundant (Gaston and Locke, 2000). In some cases, crop residues on the soil surface under conservation tillage seemed to disrupt microorganisms' activity (Locke and Harper, 1991b; Sorenson et al., 1991), and, compared with conventional tillage, lag phases in the activation of mineralisation may occur (Seifert et al., 2001b). In addition, a greater availability of carbon under conservation tillage compared with conventional tillage can defer the use of pesticides as a source of carbon and thus their degradation in soil (Locke and Harper, 1991b).

Mineralisation is considered as the last step of pesticide degradation, leading to its complete removal from the soil. The mineralisation results are highly contrasted according to pesticides, location sites and incubation conditions (Table 5) and do not allow any conclusion on an increase or a limitation of mineralisation in conservation tillage. In the same way, nonextractable residue fractions vary widely depending on pesticides, techniques and soils. However, the formation of non-extractable residues is often correlated with the degradation half-life of the molecules (Gaston and Locke, 2000; Lavorenti et al., 2003; Zablotowicz et al., 2000). Locke and Harper (1991b) have also shown that the difference in non-extractable residues of metribuzin between conventional tillage and conservation tillage was mainly due to the coarse fraction of organic matter.

Moreover, although little information is generally mentioned in the literature, the age of the tillage system seems to be a major source of differences in pesticide degradation (Wagner et al., 1996). Sadeghi and Isensee (1996) have evaluated the effect of reversing the tillage of 7-year-old no-tillage and conventional tillage field plots on atrazine degradation. Their results suggested that, after reversing a well-established tillage, time for the new tillage to fully develop its particular effects on atrazine degradation may be shorter for a new conventional tillage than for a new no-tillage. Another major source of degradation variability in field studies is the inter-annual variability of climatic conditions. It can hide or completely cancel the effects of tillage and it can lead to opposite conclusions depending on the year (Baughman et al., 1996; Mills and Witt, 1991; Sorenson et al., 1991).

In most cases, pesticide degradation forms one or several degradation products. In the same manner as for mother compounds, their degradation can be modified by tillage practices. For example, under a notillage system, an accumulation of polar metabolites formed by the degradation of metribuzin was observed, while they were degraded under conventional tillage (Locke and Harper, 1991a, b). On the contrary, the formation of alachlor metabolites was faster but their degradation was slower under conventional tillage than under conservation tillage (Locke et al., 1996). Similarly, the accumulation of a fluometuron metabolite, trifluoromethylphenylurea (TFMPU), occurred mainly under conventional tillage (Zablotowicz et al., 2000), and another of its metabolites, dimethylfluometuron, seemed to be formed mainly when the air flow conditions, due notably to tillage operations, were favourable (Locke et al., 2005).

Pesticide formulation also appeared to have important consequences on their persistence. For example, whatever the tillage system, starch encapsulation of atrazine and alachlor tended to increase their field persistence (Gish et al., 1994).

Spatial variability of local conditions seems to be an important source of pesticide degradation variation. In their study, Gaynor et al. (1987) observed that the ridge tops retained more herbicide than the furrows, resulting in differences in pesticide persistence. Recently, Alletto et al. (2008) have examined the degradation of the diketonitrile metabolite of isoxaflutole under two tillage systems. Under conventional tillage, the herbicide persistence was found to be highly variable vertically and laterally according to soil sample location within the tilled horizon. The main source of variation in this case was the tillage operation with the mouldboard plough. Under mulch tillage with disk harrowing,  $MT_{Dk}$  (Table 1), the effect of tillage was minimised but variations in diketonitrile persistence were associated with the vertical distribution of organic carbon. Both of these studies highlight the importance of the soil sampling strategy in order to provide accurate assessment of environmental impacts of agricultural practices. Last, the mulch can reduce water content and temperature variations (Bragagnolo and Mielniczuk, 1990; Unger, 1987), leading to a lower variability of degradation under conservation tillage compared with conventional tillage (Mills and Witt, 1991).

#### 3.3 Summary and Recommendations

Results about tillage effects on pesticide degradation are highly contrasted. Studies of the effects of management practices on soil properties and microorganism activity have to deal with complex interactions between soil physics, physico-chemistry and microbiology which control the microbial activities involved in pesticide breakdown. Particularly the strong coupling between retention and degradation processes highly depends on soil physical and hydrodynamic properties, ensuring the access of microbes to water, substrates and the movement of solutes such as pesticides to sorption and degradation sites. Degradation studies should therefore pay great attention to the soil conditions (dynamics of temperature, water content, pH, N content, etc.), which have been shown to be largely modified by the presence of the mulch at the soil surface. In the same way, studies focusing on the understanding and prediction of pesticide degradation in soil should use dynamic experimental systems where soil solution is allowed to move and be renewed (e.g. soil columns) rather than static (batch) experimental systems. Improving knowledge and quantitative prediction of pesticide degradation in soils also implies an accurate estimation of the spatial distribution and temporal dynamics of active degradative populations. Research efforts should thus be oriented at (a) improving field study for a better understanding of the in situ dynamics of these microorganisms and their location ('hot-spots'), and (b) establishing a comprehensive link between this dynamics and agricultural operations. Particular attention should be paid to the soil sampling strategy in this regard.

# 4 Transfer of Pesticides

Depending on their mobility and their persistence, pesticides can migrate within and outside the soil and contaminate other compartments of the environment, such as water and air. The three main transfer processes are volatilisation, leaching and runoff. The relative importance of each of these processes depends on the application conditions, the pesticide properties, the climatic conditions and the soil properties partly governed by agricultural practices.

## 4.1 Volatilisation

Volatilisation is an important pathway for the loss of pesticide that is controlled by the pesticide properties (such as saturated vapour pressure, Henry constant,  $K_{OC}$ ), the soil properties (temperature, water content, organic carbon content), the farming operations (mode of application, soil roughness, presence of a mulch) and the climatic conditions (wind, solar radiation, temperature) (Bedos et al., 2002). Although volatilisation is often mentioned in the literature to explain differences in the pesticide persistence due to tillage operations (Banks and Robinson, 1982; Curran et al., 1992; Gaynor et al., 2000), this phenomenon remains poorly studied. Whang et al. (1993) highlighted a higher volatilisation in conservation tillage than in conventional tillage that was attributed to the presence of a mulch. In their study, four days after treatment, transfer by volatilisation accounted for 48 and 18% of applied fonofos, 23 and 7% of applied chlorpyrifos and 0.9 and 0.7% of applied atrazine under no-tillage and conventional tillage, respectively. Wienhold and Gish (1994) also observed a larger volatilisation of alachlor and atrazine under conservation tillage, but only until the first rainfall occurred (5 days after treatment). Following this rainfall, volatilisation in conservation tillage was strongly slowed down, which could be explained by the migration of the pesticides from the mulch to the soil. After 35 days, the cumulative loss accounted for 9 and 14% of applied alachlor and 4 and 9% of applied atrazine under conservation tillage and conventional tillage, respectively. In addition, this study showed a significant effect of pesticide formulation on volatilisation. Starch-encapsulated alachlor was less volatilised under conservation tillage than the commercial formulation, and a similar effect was measured for atrazine under both conservation tillage and conventional tillage systems. Volatilisation directly depends on environmental conditions and is favoured first by high temperatures (Glotfelty, 1987; Weber et al., 2002) and also by wet soils that keep the molecules available in water solution. Weber et al. (2006) observed during a year of monitoring that volatilisation reached 22 and 32% of applied metolachlor in conservation tillage and conventional tillage, respectively. Due to the mulch, soil surface water content was higher under conservation tillage, which slowed down the soil warming and thus reduced the amount of herbicide lost.

### 4.2 Leaching

Pesticide properties play a decisive role in the modifications of the leaching risk by tillage. First, retention properties determine the mobility of the molecules (Singh N et al., 2002) and directly influence their transfer to groundwater. Masse et al. (1998) reported that leaching of atrazine and deethylatrazine was more significant under conservation tillage, whereas metolachlor leaching, that has a stronger sorption capacity, was not affected by tillage. Water solubility of pesticides also influences their leaching. As an example, alachlor and cyanazine leaching was related to the interaction between tillage and water solubility (Sadeghi and Isensee, 1997). Under no-tillage, the migration depth of alachlor in soil was lower than that of cyanazine, whereas the opposite was observed under conventional tillage. The highest solubility of alachlor compared with that of cyanazine allowed a migration through the soil matrix, thus increasing the possibilities of adsorption, while cyanazine rather circulated via macropores under conservation tillage. Water solubility of the molecules may be modified by soil pH, which could increase leaching risk (Li et al., 2003). For sulfentrazone, solubility in water was, for example, multiplied by a factor of 16 when soil pH rose from 6 to 7.5 (Reddy and Locke, 1998). Other studies have shown a positive correlation between the half-life of several molecules (alachlor, atrazine, cyanazine, metolachlor, metribuzin and simazine) and their concentrations measured in drains without any tillage effect on these transfers (Logan et al., 1994; Ritter et al., 1996).

Formulation, by modifying solubility in water, persistence and retention in soils of pesticides, also influences their transport. Micro-encapsulation of alachlor increased its solubility in water and its transport through the soil matrix (Sadeghi et al., 1998). Starchencapsulation of atrazine limited its losses by leaching in both no-tillage and conventional tillage (Gish et al., 1994, 1995). Hall et al. (1998) also observed a good efficiency of starch-encapsulation of atrazine, allowing a reduction of the leaching losses by a factor of 2 to 4 compared with the commercial formulation, depending on the year and on the application rate. Likewise, coating of metolachlor with a polyurea polymer helped reduce losses by a factor of 2 compared with the commercial formulation (Hall et al., 1998). According to these results, an efficient control of leaching seems achievable via adequate pesticide formulations.

Pesticide leaching depends on soil physical properties, such as the hydraulic conductivity, which is directly influenced by the soil structure created by tillage. Many studies on pesticide leaching have been conducted under field conditions, on plots equipped with ceramic cups, lysimeters or drains (Table 6), and under laboratory conditions, on undisturbed soil columns with simulated rainfalls (Table 7). Although most of these works showed greater losses under conservation tillage (Isensee and Sadeghi, 1997; Isensee et al., 1990; Masse et al., 1998; Singh et al., 2002; Weber et al., 2006), results are contrasted, some of them indicating no effect of tillage (Clay et al., 1998; Fomsgaard et al., 2003; Gaynor et al., 2000; Granovsky et al., 1993; Weed et al., 1995), or even greater losses under conventional tillage (Düring and Hummel, 1993; Gish et al., 1995; Levanon et al., 1993).

On one hand, mulch on the soil surface absorbs rainfall energy, thus avoiding the formation of soil crust (Baumhardt and Lascano, 1996; Blevins and Frye, 1993) and pore sealing (Ela et al., 1992). On the other hand, the continued deposition of crop residues on the soil surface appears to contribute to macropore development by stimulating earthworm activity (Bouché, 1972; Edwards et al., 1988; Rovira et al., 1987; Satchell, 1983). Under conservation tillage, and more particularly under no-tillage, the macropore network formed by earthworm burrows, root channels and cracks is not disrupted by tillage and thus may allow downward flows of water and solutes at a higher rate than if movements occurred only through the soil matrix. Because of this bypass of the soil matrix, this type of water and solute movement is called 'preferential flow through macroporosity'. It occurs mainly during saturated conditions. In the soil matrix, solutes move by convection-dispersion and, due to significant contact between the liquid and solid phases, opportunities for pesticide retention are greater than in macropores (Shipitalo and Edwards, 1996). Ogden et al. (1999) showed that soil tillage destroyed the connectivity of the macropores and thus increased fluxes through the soil matrix.

Proportionally to their quantity, crop residues can limit losses due to leaching. By reducing the infiltration rate at the soil surface, i.e. at the crop residues/ soil interface, the mulch promotes pesticide fluxes within the soil matrix, thus avoiding preferential flows through macropores (Sigua et al., 1993). On the other hand, at the residues/soil interface or when residues

able o Field studies c	or pesuicide reacriming (		ose) under		soil propertic	Sč			
			Danth	Water	4	Organic		Leaching	
Pesticide	Lose kg a.i. ha <sup>-1</sup>	Tillage <sup>1</sup>	cm	sampung system	Clay	g kg <sup>-1</sup>	Hq	% applied dose	Reference
Alachlor	2.2	NT T d	0-120	I	I	17.6.20.8	5.7-7.0	$0.0002 - 0.10^2$	(Weed et al., 1995)
		MT 2		1	1	186-213	57-68		
		CT		I	I	19.7–20.4	6.3-6.6		
Atrazine	$2.2, 1.5, 1.8, 1.9^3$	NT	0-100	$\mathrm{D}^4$	I	Ι	I	$0.09, 0.08, 0.15, 0.05^3$	(Masse et al., 1996)
Desethylatrazine		CT			I	I	I	0.04, 0.02, 0.12, 0.02	
		LN E			I	I	I	0.10, 0.06, 0.19, 0.09	
		CL			I	I	I	0.07, 0.03, 0.15, 0.04	
Atrazine	2.8	NT	0-120	I	I	17.3–21.4	5.7-7.0	0.02 - 0.35	(Weed et al., 1995)
		RT		I	I	17.6–20.8	5.9-6.7		
		$MT_{Ch}$		I	I	18.6–21.3	5.7-6.8		
		CI		I	I	19./20.4	0.0-2.0		
Atrazine	1.3	$\mathrm{MT}_{Ch}$ CT	1	1 1	120–300 120–300	15.4–24.2 15.7–23.5	5.4–4.9 5.3–5.8	0.07-0.11, 0.08-0.22 0.03-0.07, 0.11-0.12	(Fortin et al., 2002)
Atrazine	1.7	NT CT	0-120	$L^4$	324-421	2.0-12.0	5.5-6.7	0.15-0.86, 0.21-9.60 <0.01-0.19, 0.75-0.85	(Hall et al., 1989)
Atrazine, Cyanazine Simazine	1.7, 2.2, 1.7	MT mulch CT	0-120	Г	1 1	1 1	1 1	3.0–5.1 0.69–0.93	(Watts and Hall, 1996)
Cyanazine	2.2	NT CT	0-120	Г	324-421	2.0-12.0	5.5-6.7	0.03-0.23, <0.10-4.73 <0.01-0.15, 0.32-0.56	(Hall et al., 1989)
Dicamba	0.56	NT	I	I	I	2.0-12.0	5.5-6.7	1.99, 0.39, 5.56, 1.05, 2.45	(Hall and Mumma, 1994)
		CT	I	I	I			<ol> <li>2.00</li> <li>&lt;0.01, 0.20, 0.20, 0.0, 0.0, 0.58</li> </ol>	
Fluometuron	1.6	NT	06-0	L	130	1.3–17.7	5.61	29.9–37.4, 37.9–50.8, 0–10.3, 0.8–54.7	(Essington et al., 1995)
		CT				1.1-10.2	5.85	$11.9-69.3,  30.0-79.5, \\ 6.1-28.7,  3.4-73.3$	
Glyphosate AMPA	0.8	NT (20) <sup>5</sup> NT	0-110	L	134–227	1.1–19.2	5.9-7.6	0.022 0.018	(Fomsgaard et al., 2003)
Glyphosate AMPA		CT CT			83–268	0.5-11.3	6.4–8.7	0.0305 0.0205	
									(continued)

809

					Soil propertie	s			
				Water		Organic		Leaching	
	Dose		Depth	sampling		carbon		% applied	
Pesticide	kg a.i. ha <sup>-1</sup>	Tillage <sup>1</sup>	cm	system	Clay	g kg <sup>-1</sup>	рН	dose	Reference
Metolachlor	2.6, 2.6, 2.6, 2.4	NT	0-100	D	I	1	I	0.00, 0.02, 0.02, 0.04	(Masse et al., 1996)
		CT			Ι	I	I	0.00, 0.00, 0.02, 0.01	
Metolachlor	2.2	MT mulch	0-120	Γ	I	I	I	2.46	(Watts and Hall, 1996)
		CT			Ι	Ι	I	0.37	
Metolachlor	2.5	$\mathrm{MT}_{Ch}$	I	I	120-300	15.4-24.2	5.4-4.9	0.02 - 0.04, 0.06 - 0.12	(Fortin et al., 2002)
		CT	Ι	Ι	120–300	15.7–23.5	5.3-5.8	0.01 - 0.03, 0.07 - 0.10	
Metolachlor	4.48	NT (10)	79-07	L	60–290	3.0-6.4	4.4–6.1	1.4, 6.7	(Weber et al., 2006)
		CT.						0.7, 4.4	
Metolachlor	2.2	NT	0-120	L	324-421	2.0-12.0	5.5-6.7	0.01-0.47, < 0.10-4.19	(Hall et al., 1989)
		CT						<0.01-0.10, 0.25-0.61	
Metribuzine	0.45	NT	0-120	I	I	17.3–21.4	5.7-7.0	0.14 - 0.87	(Weed et al., 1995)
		RT		Ι	I	17.6–20.8	5.9-6.7		
		$\mathrm{MT}_{Ch}$		Ι	Ι	18.6–21.3	5.7-6.8		
		CT		I	I	19.7–20.4	6.3-6.6		
Simazine	1.7	NT	0-120	L	324-421	2.0-12.0	5.5-6.7	0.06-1.76, 0.18-8.36	(Hall et al., 1989)
		CT						0.01-0.18, 1.50-1.63	
<sup>1</sup> Codes of tilla	ge practices are repor	ted in Table 1.	-						
<sup>-</sup> Hypnens are t <sup>3</sup> Commas are t	used to indicate the ra- used to separate differ	nge of variation of ent vears of stud	or values. v.						
<sup>4</sup> D: water sam	pling in drains; L: wa arenthesis indicate the	ter sampling in ly e age in years of	/simeters. the tillage s	ystem.					

Table 6 (continued)

L. Alletto et al.

Table 7 Labora	tory studies on pe	sticide leaching	(% of applied	l dose) under	r different tillage prac	tices			
				Soil propert	ies		Rainfall		
Pesticide	Dose kg a.i. ha <sup>-1</sup>	Tillage <sup>1</sup>	Depth cm	Clay	Organic carbon g kg <sup>-1</sup>	Hq	intensity $\min h^{-1}$	Leaching % applied dose	Reference
Alachlor	2.2	NT (16) <sup>2</sup> CT	0–30	1 1	1 1	1 1	50	1.6 0.4	(Weed et al., 1998)
Alachlor	3.3	NT (8) CT	0-10	1 1	1 1	1 1	50	29.7 14.4	(Clay et al., 1991)
Atrazine	2.8	CT CT CT	0–18	170	$\begin{array}{c} 10.2 - 15.3^{3} \\ 11.8 - 12.6 \\ 10.2 - 15.3 \\ 11.8 - 12.6 \end{array}$	6.4-5.9 6.4-6.7 6.4-5.9 6.4-6.7	1 1 1 1	14.7 19.3 12.9 20.4	(Levanon et al., 1993)
Atrazine	1.3	K	0-10		NT: 7.5-24.4 CT: 7.0		~ ~ ~ ~ ~ ~ ~ ~ ~ ~	58.4 42.7 47.2 33.2 31.6 29.4 20.1	(Sigua et al., 1995)
Atrazine	1.3	5 5 5 5 5 5 5 5 5 5 5 5	0-10		NT: 24.4–7.5 CT: 7.0	1 1 1 1 1 1 1 1 1	0 0 0 0 0 0 0 0 0	50.7 35.7 34.9 40.3 38.2 37.0	(Sigua et al., 1995)
Carbofuran	1.9	CT NT CT NT	0-18	170	10.2–15.3 11.8–12.6 10.2–15.3 11.8–12.6	6.4–5.9 6.4–6.7 6.4–5.9 6.4–6.7	1 1 1 1	4.5 18.7 4.2 7.7	(Levanon et al., 1993)
Diazinon	2.5	CT NT CT NT	0-18	170	10.2–15.3 11.8–12.6 10.2–15.3 11.8–12.6	6.4–5.9 6.4–6.7 6.4–5.9 6.4–6.7	1 1 1 1	0.25 0.35 0.0 0.0	(Levanon et al., 1993)

811

Table 7 (continue	(p								
				Soil propert	ies		Rainfall		
	Dose	-	Depth	6	Organic carbon	:	intensity	Leaching	c k
Pesticide	kg a.i. ha <sup>_1</sup>	Tillage	cm	Clay	g kg_'	Hd	mm h_'	% applied dose	Reference
Metamitron	4	NT (20)	0-30/40	112–210	9.0–18.0		0.16, 0.38, 0.46	0.03, 3.0, 10.1	(Düring and Hummel, 1999)
		CT		151-217	9.0 - 1.0			0.3, 5.0, 12.2	
Metolachlor	2.25	LN	0-18	170	10.2 - 15.3	6.4-5.9	I	8.2	(Levanon et al., 1993)
		CT			11.8-12.6	6.4–6.7	I	10.2	
		LN			10.2 - 15.3	6.4-5.9	I	6.5	
		CT			11.8-12.6	6.4–6.7	I	9.3	
Metolachlor	10	NT (20)	0-30/40	112-210	9.0–18.0	I	0.25, 0.42	1.8, 5.7	(Düring and Hummel, 1999)
		CT		151–217	9.0-1.0	I		3.6, 12.4	
Metolachlor	20	NT CT	0-15	280	19.0	7.3	saturation	38 27	(Singh N et al., 2002)
Terbuthylazine	5	NT (20)	0-30/40	112-210	9.0–18.0	I	0.25, 0.42	3.5, 9.7	(Düring and Hummel, 1999)
		CT		151–217	9.0-1.0	I		5.2, 12.2	
Terbuthylazine	10	NT CT	0-15	280	19.0	7.3	saturation	11 6	(Singh N et al., 2002)
<sup>1</sup> Codes of tillage	practices are rep	orted in Table	1. .4						

 $^2$  Numbers in parenthesis indicate the age in years the tillage system.  $^3$  Hyphens are used to indicate the range of variation of values.

L. Alletto et al.

and soil are mixed, the presence of these two compartments that have different water-holding capacities and hydraulic conductivities can also generate preferential flow (Kasteel et al., 2007; Ma and Selim, 2005).

There is no direct relationship between the number of macropores and the intensity of preferential flow, since some macropores do not participate in the conduction of water (Shipitalo et al., 1990). Preferential flow activation seems to depend, in particular, on the initial soil water content, but in contrasted ways according to studies (Granovsky et al., 1993; Shipitalo and Edwards, 1996). Although preferential flows through macropores occur in both conservation and conventional tillage (Andreini and Steenhuis, 1990; Essington et al., 1995; Gish et al., 1991; Granovsky et al., 1993), this dependency on initial water content was found to be greater under conservation tillage (Flury et al., 1995; Sigua et al., 1995). At low initial water content, the hydrophobicity of organic materials at the soil surface under conservation tillage could limit the entry of water into the soil matrix, creating locally and temporarily saturation conditions, thus favouring macropore fluxes (Edwards et al., 1989, 1992a; Phillips et al., 1989; Shipitalo et al., 1990). Despite earthworm burrows being found to have high pesticide sorption capacities due to high amounts of organic compounds in burrow linings (Edwards et al., 1992b; Stehouwer et al., 1993), water and pesticide transfer rates through macropores are generally higher than for the soil matrix (Shipitalo and Edwards, 1996). Other studies showed a higher leaching of atrazine (Kitchen et al., 1998; Seyfried and Rao, 1987; Sigua et al., 1995) and fluometuron (Essington et al., 1995) under wet soil initial conditions. A better understanding of the effect of the initial water content on the dynamics of macropore flow is thus needed to improve the control of pesticide leaching and to complement modelling efforts. An illustration is given by Sigua et al. (1995), who observed under laboratory conditions that a decrease in soil matric potential from -1kPa (near saturation) to -33 kPa led to a decrease of 15% in atrazine leaching.

In addition, the temporal dynamics of preferential flow through macropores needs to be taken into account. During wetting, the number of macropores hydraulically active tends to increase, thus increasing the possibilities of preferential flow (Jaynes et al., 2001; Kung et al., 2000; Malone et al., 2001). However, under unsaturated conditions, macropores are inactive but still the observed movements of solutes through the soil matrix may be faster than those estimated by the convection-dispersion equation.

To describe solute movements in the soil matrix better, the Mobile–Immobile water Model (MIM) has been developed (Coats and Smith, 1964). This model considers that the water-filled pore space is partitioned into two domains: a mobile domain where water can move and solute transport is due to convection and dispersion, and an immobile domain where water is stagnant and solutes move only by diffusion (Coats and Smith, 1964; van Genuchten and Wierenga, 1976). In this model, it is possible to distinguish different degradation and retention kinetics between the two domains (van Genuchten and Wagenet, 1989), allowing a better description of preferential fluxes under unsaturated conditions (Gaston and Locke, 1996; Gaston and Locke, 2000; Pot et al., 2005). This type of preferential flow has been identified for pesticides under both conventional and conservation tillage (Gaston and Locke, 1996; Gaston and Locke, 2000; Singh N et al., 2002), but laboratory studies on undisturbed soil columns showed that immobile water fractions were higher under conservation tillage than under conventional tillage, with 56 vs. 49% (Singh and Kanwar, 1991) and 56 vs. 35% (Singh N et al., 2002), respectively. As for macropore fluxes, preferential flows within the soil matrix also seem to increase under wet soil conditions (Shipitalo and Edwards, 1996). Despite its strong impact on solute transfers, little data about the impact of tillage on MIM-type preferential flow has been published and thus further experimental studies are needed.

Intensity and timing of rainfall after treatment are major factors affecting pesticide leaching (Granovsky et al., 1993; Heatwole et al., 1997; Isensee and Sadeghi, 1994; Masse et al., 1996). Their effects may be greater than those generated by tillage (Gaynor et al., 1995; Granovsky et al., 1993; Otto et al., 1997). During small, low-intensity rainfalls, pesticides intercepted by the mulch can be washed off and then can penetrate slowly into the soil matrix where sorption processes can reduce their leaching (Shipitalo et al., 1990). On the contrary, during high-intensity rainfalls, wash-off is significant and the hydrophobicity of organic residues can slow down infiltration and activate preferential flow in macropores (Isensee and Sadeghi, 1994). By removing the mulch under conservation tillage, Sadeghi and Isensee (1997) observed that

leaching was reduced and finally, lower losses under conservation tillage than under conventional tillage were measured. Furthermore, high-intensity rainfall generally leads to greater water and solute fluxes than low-intensity rainfall (Quisenberry et al., 1994; Trojan and Linden, 1992), with most pesticide leaching occurring during the first 2 or 3 rains (Fortin et al., 2002). For atrazine, Sigua et al. (1993) showed on undisturbed soil columns collected under no-tillage plots that the intensity of rainfall determines the percentage of losses: 33% of the applied dose for an intensity of 3 mm  $h^{-1}$  and 52% at 9 mm  $h^{-1}$ . In addition, if a small and low-intensity rainfall (1.5 mm for Sadeghi and Isensee, 1997, or 5 mm for Shipitalo et al., 1990) preceded a leaching event, then pesticide transport could be reduced by 50% compared with treatments that did not receive this preliminary rain. Moreover, these studies also showed that allowing time for sorption and diffusion in the soil matrix to occur can reduce pesticide movements in soils.

Some questions remain about how rapidly soil hydraulic properties change and preferential flow paths develop when tillage is modified. Very few data have been published on this aspect. In their study, Isensee and Sadeghi (1996) reversed the tillage of 7-year-old no-tillage and conventional tillage plots on which preferential flows had been previously observed (Isensee and Sadeghi, 1994; Isensee et al., 1990; Sadeghi and Isensee, 1992). They found that, immediately after ploughing, preferential flows under the new conventional tillage had disappeared and that sufficient macropore pathways were developed under the new no-tillage in only one year without tillage, leading to significant leaching differences. However, it seemed that two years were required before preferential transport systems became well established. Moreover, in the new conventional tillage, the macropore network inherited from 7 years of no-tillage and situated below the newly ploughed horizon appeared to be functional and preferential flow could still occur (Isensee and Sadeghi, 1997). Soil structural stability, which depends on soil texture, also greatly influences the dynamics of preferential flow (Singh et al., 2002). For example, in four soils of various textures under no-tillage, clay content enhanced macropore stability, leading to a leaching of atrazine 40% higher than for a sandy soil (Sadeghi et al., 2000).

Most of the studies about the effects of tillage on leaching compare contrasted techniques such as ploughing and no-tillage. Other conservation techniques such as mulch tillage or ridge tillage are still poorly studied in terms of pesticide leaching. In some studies of mulch tillage systems, greater leaching of atrazine, cyanazine, simazine and metolachlor was measured compared with conventional tillage (Watts and Hall, 1996). To the contrary, other studies did not find any differences in atrazine and metolachlor leaching between mulch tillage and conventional tillage, concluding that the differentiation between these systems was too weak (Fortin et al., 2002). Between mulch tillage and ridge tillage systems, differences in the migration of atrazine were also poorly marked (Kitchen et al., 1998). However, localisation of the herbicide in seed rows under ridge tillage allowed a decrease by a factor of 3 in the applied doses and could significantly reduce losses by leaching (Lamb et al., 1998; Lowery et al., 1998). In their study comparing three different tillage systems, Mazzoncini et al. (1998) observed a deeper migration of diclofop-methyl after a heavy rainfall under no-tillage, while the pesticide was found mainly in the seedbed layer (0-5-cm depth) under mulch and conventional tillage.

## 4.3 Runoff

Agricultural runoff is the primary mechanism contributing to pesticide contamination of surface waters. The main objective of conservation tillage is to reduce runoff and soil erosion (Gebhardt et al., 1985).

Runoff is also affected by pesticide properties. Sorption properties directly act on the dominant mode of transport. For pesticides that have a high sorption capacity on organo-mineral particles, such as glyphosate, trifluralin, paraquat or organochlorine pesticides, surface transport is associated with erosion and soil particle transport (Wauchope, 1978). Potter et al. (2004) indicated that 55% of pendimethalin ( $K_{OC}$  = 5000 L kg<sup>-1</sup>) losses were bound to sediment. However, for most pesticides, transport to surface water is realised in solution in water runoff (Wauchope, 1978). Losses in solution represented from 88 to 97% of total losses of alachlor and cyanazine (Hansen et al., 2001), 99.8% of atrazine losses (Basta et al., 1997) and, according to tillage practices, from 89 to 98% of fluometuron losses (Potter et al., 2004). For herbicides transported only in solution, such as chlorimuron and

nicosulfuron, the interest of conservation techniques, especially no-tillage used to control erosion, is very limited (Afyuni et al., 1997). For some authors, however, pesticide transport in solution may be overestimated due to desorption from suspended sediments during runoff or sample storage and thus the importance of the solid phase as a source of pesticides in runoff would be underestimated (Hansen et al., 2001).

Logan et al. (1994) compared the effects of conventional tillage and no-tillage on the transport of atrazine, alachlor, metolachlor and metribuzin by runoff. No difference between tillage systems was found but runoff losses were positively correlated with the halflives of the molecules. In other studies, water solubility directly determines the loss by runoff and explains differences in behaviour between alachlor and cyanazine (Hansen et al., 2001) or between atrazine and metribuzin (Gaynor et al., 2001). Isoxaflutole has a low solubility in water but its hydrolysis forms a diketonitrile metabolite (active ingredient) which is 50 times more soluble and thus transported by runoff (Rector et al., 2003). However, mainly because of analytical costs, metabolite monitoring in runoff is not systematic even in research studies. For fluometuron, Potter et al. (2004) indicated that 50% of the herbicide losses occurred as desmethylfluometuron metabolite. Results of runoff losses for some molecules, such as atrazine or fluometuron, are often contrasted under conservation tillage despite similar water solubility and retention by organic compounds. Potter et al. (2003) suggested a relationship to predict the mode of transport of pesticides (in solution or adsorbed) based on their K<sub>OC</sub> and their solubility.

Pesticide solubility may vary according to formulation. For example, micro-encapsulation of alachlor increases its solubility, allowing a faster migration within the soil matrix and a decrease in its runoff compared with the commercial formulation (Isensee and Sadeghi, 1993). Metolachlor solubility is higher than that of atrazine but it may be reduced by its formulation, which may then limit its runoff compared with atrazine (Sadeghi and Isensee, 2001).

Concerning soil properties, increasing soil organic matter content in topsoil improves soil aggregate stability and cohesion, leading to a significant decrease in soil loss (Rhoton et al., 2002). It appears, however, that tillage effects on pesticide transfer by runoff or erosion are, as for leaching, mixed (Table 8). Many works highlighted a decrease in pesticide losses by runoff under conservation tillage due to a decrease in water runoff volume as the degree of tillage is reduced (Seta et al., 1993; Tebrügge and During, 1999; Watts and Hall, 1996; Webster and Shaw, 1996). However, in some studies, pesticide concentrations in runoff from conservation tillage plots, especially no-tillage plots, were higher than in runoff from conventional tillage plots. As a consequence, conservation tillage plots may generate greater pesticide losses despite a lower water runoff volume (Heatwole et al., 1997; Kenimer et al., 1987; Shipitalo and Owens, 2006; Webster and Shaw, 1996). In other studies, water runoff volume was higher under conservation tillage (Gaynor et al., 1995; Myers et al., 1995). A review of tillage effects by Fawcett et al. (1994) concluded that, under natural rainfall, conservation tillage was efficient in controlling runoff, erosion and pesticide losses. Under simulated rainfall, the amount and intensities of water applied shortly after treatment resulted in very mixed results in conservation tillage. For Fawcett et al. (1994), intensive rainfall was more an indicator of the limits of the efficiency of conservation tillage to control runoff rather than an accurate representation of their effects under real field conditions. Modelling was used to identify and classify the most suitable practices to control atrazine runoff in a watershed (Harman et al., 2004). In this study, conservation tillage was found to be marginally effective. The most effective practices were: sediment retention ponds, grass filter strips, atrazine application at planting time in bands and wetlands construction. According to these contradictory results, it seems necessary to clarify the main conditions and mechanisms involved in pesticide transfer by runoff that may lead to an efficient or inefficient control of losses by conservation tillage.

Rainfall is the most important parameter controlling water runoff, soil erosion and pesticide loss. Depending on its arrival time after treatment, its intensity and the interval between two rainfall events, rainfall can lead to very contradictory results for the same study site, the same molecule or even the same practice (Baker and Johnson, 1979). Pesticide transport in soils directly depends on the interval between treatment and the first rainfall (Baker and Johnson, 1979; Potter et al., 2004; Shipitalo and Owens, 2003). Several studies indicated that conservation tillage was inefficient in controlling runoff when heavy rainfall occurs quickly after pesticide application (Rector et al., 2003; Shipitalo and Owens, 2003; Shipitalo and Owens, 2006). Moreover,

Table 8 Field st	tudies of	water and I	pesticic	de runoff under simulated	d (S) (	or natura	l (N) rainfall under dil	fferent	tillage practices		
			Soil p	properties					Runoff		
Dose Pesticide kg a.	e i. ha <sup>-1</sup>	Tillage <sup>1</sup>	Clay	Organic carbon (OC) g kg <sup>-1</sup>	Hq	Slope %	Rainfalls mm	Type	water % applied dose	pesticide % applied dose	Reference
Alachlor 2.24		NT RT CT	1 1 1	5.8–17.4 <sup>2</sup>	1 1 1	12-18 <sup>2</sup>	657, 487, 463, 425 <sup>3</sup>	$\rm N^4$	$\begin{array}{c} 13,4,12,8^3\\ 17,4,8,8\\ 17,8,18,9\end{array}$	$\begin{array}{c} 7.1,0.45,1.8,0.1^3\\ 8.0,0.8,0.45,<0.1\\ 14.3,0.7,4.1,0.2 \end{array}$	(Baker and Johnson, 1979)
Alachlor 2.80		NT MT <sub>Ch</sub> CT	220	17.4	I	9	136	Z	34, 41 28, 54 35, 54 44, 48	0.32, 4.8 0.25, 2.1 0.36, 6.0 0.29, 3.0	(Sauer and Daniel, 1987)
Alachlor 1.79		NT CT	1 1	7.4–23.7 6.6–6.9	1 1	3-5	1 1	I I	$4-6, 3-4^1$ 5-6, 3-5	0.22-0.24, 0.15-0.24 0.48-0.57, 0.15-0.28	(Isensee and Sadeghi, 1993)
Alachlor 3.36		NT C <sup>5</sup> NT U/D <sup>5</sup> CT C CT U/D	1 1 1 1	12.8		7–11	67 66 56 56	$\mathbf{S}^4$	19 34 58	0.06 0.08 0.1 0.2	(Felsot et al., 1990)
Alachlor 3.36		NT C <sup>5</sup> NT U/D <sup>5</sup> ST C ST U/D RT C RT U/D MTC <sub>0</sub> /C CT C CT C CT U/D		12.8		8 8 9 9 9 9 8 8 10 10 10 10 10 10 10 10 10 10 10 10 10	63 64 63 63 64 64 63 64 64	S	6 23 29 29 29 29 29 29 29 20 20 20 20 20 20 20 20 20 20 20 20 20	0.09 2.1 0.06 0.75 0.75 1.3 1.7 2.3 6.5	(Felsot et al., 1990)
Alachlor 2.24-	-3.36	$\begin{array}{c} \mathrm{NT} \ 1\\ \mathrm{NT} \ 2\\ \mathrm{MT}_{Ch} \ 1\\ \mathrm{MT}_{Ch} \ 2\end{array}$	1 1 1 1	1 1 1 1		11 10 13	629–1012 686–1025 609–981 663–1007	Z	1-10 2-14 <0.01-1 1-15	Tr-0.23 0.01-0.19 Tr-<0.01 Tr-0.22	(Shipitalo et al., 1997)
Alachlor 2.2 2.2 0.7 L 0.7 L	°oc <sup>o</sup> oco	MT <sub>Ch</sub> RT CT MT <sub>Ch</sub> CT	1 1 1 1 1 1	14.0 17.4 13.4 14.0 17.4 13.4	7.1	8-10	240, 480	Z	1 1 1 1 1 1	0.4, 0.2 0.1, 0.1 1.0, 0.6 0.1, 0.06 0.1, 0.06 0.2, 0.06	(Hansen et al., 2001)

(Shipitalo and Owens, 2006)	(Watts and Hall, 1996)	(Heatwole et al., 1997)	(Shipitalo and Owens, 2003)	(Baker and Johnson, 1979)	(Sauer and Daniel, 1987)	(Ghidey et al., 2005)	(Isensee and Sadeghi, 1993)	(Hall et al., 1991)	(continued)
0.28 0.11 0.12	0.06, - ( -, 0.28 0.18, 0.77	0.7 0.1	0.027-0.82 0.049-4.95 Tr-0.011 0.0045-2.62 0.034-0.64 0.0010-0.59 0.0043-6.37	7.1, 0.75 19.2, 1.8 20.1, 0.09	0.32, 7.3 0.21, 3.0 0.34, 8.7 0.25, 4.3	$\begin{array}{c} 2.5, 1.9, 1.0, \\ 5.1, -, 2.0 \\ 1.1, 0.3, 0.6, \\ 1.5, 4.0, \\ 2.1 \end{array}$	1.39–1.51, ( 0.56–0.90 0.81–0.97, 0.21–0.47	-, 0.10, <0.01, <0.01 0.36, 0.33, 0.02, <0.01	
8 7 11	1 1 1	1 1	4-7 6-15 0.1-2 9-18 7-17 4-8 10-22	13, 4 17, 4 18, 8	34, 41 28, 54 35, 54 44, 48	5, 6, 7, 18, -, 1 $1, 4, 7, 18, 1$ $18, 1$	4-6, 3-4 5-6, 3-5	-, 0.1, <0.01, -, 0.01, 0.02 0.3, 0.3, 0.03, 0.04	
Z		1 1		Z	1 1 1 1		I I	1 1	
976 953 960	1 1 1	1 1		657, 487	136	941, 1158, 824, 926, 1029, 860	1 1	I I	
10-11 7-13 6-9	3-5	7	$\begin{array}{c}11\\1\\6\\6\\6\end{array}$	12–18	9	1 1	3-5 5	°−° °5 4 °5	
		5.64 5.91			I	1 1		6.7	
	1 1 1	9.7 4.4		5.8-17.4	17.4	1 1	7.4–23.7 6.6–6.9	12.0	
1 1 1	1 1 1	67 59		1 1 1	220	1 1	1 1	1 1	
$\substack{MT_{Ch}}{MT_{Dk}}$	NT MT mulch CT	NT CT		NT RT CT	NT MT <sub>Ch</sub> CT	TM	NT CT	CT CT	
1.68 3.36	1.7	1.68	2.24 2.24 2.24 1.12 1.12 1.12 1.12	2.24	2.80	2.24	1.34	1.7	
Alachlor <sup>7</sup>	Atrazine	Atrazine	Atrazine (+ DEA + DIA)	Atrazine	Atrazine	Atrazine	Atrazine	Atrazine	

Table 8 (contin	nued)								
			Soil properties				Runoff		
Pesticide	Dose kg a.i. ha <sup>-1</sup>	Tillage <sup>1</sup>	Organic carb Clay g kg <sup>-1</sup>	on (OC) Slc pH %	pe Rainfall mm	s Type	water % applied dose	pesticide % applied dose	Reference
Atrazine	2.24	NT MT CT	1 1 1	6	132	s	6 22 34	0.7 1.3 1.8	(Seta et al., 1993)
Atrazine	2.24	NT CT	232 21.5	- 10. 10.	.3 100 6	S	0.2 3	0.28 2.85	(Kenimer et al., 1987)
Atrazine	0.56	$\mathop{\mathrm{NT}}_{Ch}^{(1)^8}$ $\mathop{\mathrm{MT}}_{Dk}^{Ch}$	- 9.5	4.9 4	80	S		7.9 10.5 7.5	(Basta et al., 1997)
Atrazine	2.24	NT 1 NT 2 MT $_{Ch}$ 1 MT $_{Ch}$ 2		- 11 - 10 - 13 - 7	629–101 686–102 609–981 663–100	Z N	1-10 2-14 <0.01-1 1-15	<0.01-0.08 <0.01-1.10 <0.01 <0.01	(Shipitalo et al., 1997)
Atrazine	1.6	NT (2) – M. <sup>9</sup> NT – Sil. <sup>8</sup> CT	210 9.1	5.9 2-(	5 31.7-50	8. S	31–50, 35–38 31–56, 35–48 16–24, 34–47	2.2–9.4, 2.1–4.7 1.7–8.1, 1.9–2.5 0.5–1.6, 1.7–3.0	(Myers et al., 1995)
Atrazine	0.9 0.9 Inc. <sup>10</sup> 0.9 1.8	NT (4) CT CT NT (4)	- 13.4	6.0 1-7		N/N + S		4.99/10.51 1.26/2.73 2.48/7.72 2.64/7.52	(Rector et al., 2003)
Atrazine	0.55	$MT_{Ch} + Cov.C^{11}$ $MT_{Ch}$ $CT + Cov.C$ $CT$	1 – 14.5	Г	995, 688	Z	8, 7 14, 9 8, 6 12, 7	0.1, 1.8 0.4, 1.5 0.4, 0.7 0.6, 1.8	(Gaynor et al., 2001)
Atrazine <sup>7</sup>	2.24	$\substack{NT_{Ch}}{MT_{Dk}}$	1 1 1	10- - 6- <u>9</u>	-11 976 13 953 ) 960	Z	8 7 11	1.29 0.66 0.92	(Shipitalo and Owens, 2006)
Carbofuran	1.12	NT C NT U/D CT C CT U/D	- 12.8	- 7-1	11 67 66 56	S	19 34 58	0.2 0.6 0.1 1.21	(Felsot et al., 1990)
Chlorimuron	0.014	NT CT	80 5.3	5.9 0-	-	I	I	2.0 1.3	(Afyuni et al., 1997)
Chlorpyrifos	1.34	NT MT <sub>Ch</sub> RT CT	220 17.4	- 9	136	I	34, 41 28, 54 35, 54 44, 48	0.04, 0.10 0.05, 0.29 0.06, 0.28 0.08, 0.20	(Sauer and Daniel, 1987)

(Baker and Johnson, 1979)	(Isensee and Sadeghi, 1003)		(Hall et al., 1991)		(Hansen et al., 2001)			(Watts and Hall, 1996)	(Hall and Mumma, 1994)		(Potter et al., 2003)	(Baughman et al., 2001)	(Potter et al., 2006)	(Baker and Johnson, 1979)	(continued)
4.5, 0.2 2.0, 0.1 8.0, 0.3	1.30–1.91, 0 50–0 68	0.52-0.84, 0.15-0.27	-, 0.06, < 0.01, < 0.01	0.13, 0.30, 0.01, <0.01	2.0, 0.9 0.38, 0.8	4.4, 1.0 0.8, 0.2	0.6, 0.2 0.8, 0.2	0.02, – –, 0.17 0.11, 0.39	-, 0.12, < 0.01, < 0.01	0.46, 0.81, <0.01, <0.01	5.0 1.6	8.0, 5.3 8.2, 3.2 9.9, 4.4	1.4 1.9 0.8 0.6	0.18, 1.3, 0.05 0.09, 0.25, nd 0.36, 1.0, 0.07	
12, 8 8, 8 18, 9	4-6, 3-4	5-6, 3-5	-, 0.1, < 0.01, -, 0.02	0.3, 0.3, 0.03, 0.03, 0.04	1 1	1 1	1 1	1 1 1	-, 0.1, < 0.01, 0.02	0.3, 0.3, 0.03, 0.03, 0.04	23 23	6 S 6	20 23 50	4, 12, 8 4, 8, 8 8, 18, 9	
Z	I		I		I			I	I		S	N + S	$\begin{array}{c} \mathbf{S}\mathbf{-}\mathbf{I}_{Var}^{12}\\ \mathbf{S}\mathbf{-}\mathbf{I}_{cons}^{12}\\ \mathbf{S}\mathbf{-}\mathbf{I}_{var}\\ \mathbf{S}\mathbf{-}\mathbf{I}_{cons}\\ \mathbf{S}\mathbf{-}\mathbf{I}_{cons}\end{array}$		
463, 425	I		I		240, 480			I	I		I	376, 480	63.8 64.3 60.0 60.0	487, 463, 425	
12–18	3-5		3-5	3-4	8-10			3-5	3-5	3-4	3-4	ε	3-4	12–18	
I	I	I	6.7		7.1			I	6.7		6.5	6.4	6.5	I	
5.8-17.4	7.4–23.7	6.9–9.9	12.0		14.0 17.4	13.4 14.0	17.4 13.4	I	12.0		5.1	18.6	5.1	5.8-17.4	
I	I	I	I		1 1	1 1	1 1	I	I		32	I	32	I	
NT RT CT	NT	CT	NT	CT	$\mathrm{MT}_{Ch}$ RT	$\operatorname{CT}_{Ch}$	RT CT	NT MT mulch CT	ΝΤ	CT	ST (1) CT	NT RT CT	ST ST CT CT	NT RT CT	
2.24	1.34		2.2		2.1 2.1	2.1 $0.7  Loc^6$	0.7 Loc 0.4 Loc	2.2	0.56		0.35	1.7	2.0	1.12	
Cyanazine	Cyanazine		Cyanazine		Cyanazine			Cyanazine	Dicamba		Dimethipin	Fluometuron	Fluometuron	Fonofos	

			Soil	properties					Runoff		
	Dose			Organic carbon (O	C)	Slope 1	Rainfalls		water	pesticide	
Pesticide	kg a.i. ha <sup>–</sup>	<sup>-1</sup> Tillage <sup>1</sup>	Clay	∕ g kg <sup>−1</sup>	6 Hq	10 1	mm	Type	% applied dose	e % applied dose	Reference
Imazaquin	0.14	SS (3) CT	660	16.9	6.4 N	- AV		I	1 1	1.22, 3.39 1.53, 3.63	(Seifert et al., 2001b)
Isoxaflutole/diketonitrile	0.11 0.11 Inc. <sup>10</sup> 0.11 0.05	NT (4) CT CT NT (4)	I	13.4	6.0 1		I	+ NN		2.82/9.24 0.93/3.35 1.63/6.75 2.35 / 6.84	(Rector et al., 2003)
Linuron	1.12	$\begin{array}{c} \mathrm{NT} \ 1\\ \mathrm{NT} \ 2\\ \mathrm{MT}_{Ch} \ 1\\ \mathrm{MT}_{Ch} \ 2\end{array}$		1 1 1 1	 	10 6 7	529–1012 586–1025 509–981 563–1007	Z	1-10 2-14 <0.01-1 1-15	<0.01-0.01 <0.01-1.04 <0.01 <0.01	(Shipitalo et al., 1997)
Linuron <sup>6</sup>	1.12	$\substack{MT_{Ch}}{MT_{Dk}}$		1 1 1		$\begin{bmatrix} 0-11 \\ -13 \\ 5-9 \end{bmatrix}$	976 953 960	Z	8 7 11	0.70 0.29 0.21	(Shipitalo and Owens, 2006)
Metolachlor	3.4	$NT - S/W^{1}$ $NT - Soy^{15}$ $CT - Soy^{12}$ $CT - S/W$ $CT - Soy$	۱	18.6	6.4 3	~	213, 191, 188	N+S	8, 7, 9 -, 14, 18 9, 9, 15 9, 5, 10	$\begin{array}{c} 0.4,0.2,0.6\\ -,1.4,0.3\\ 0.8,1.1,0.6\\ 0.4,0.5,0.8\end{array}$	(Webster and Shaw, 1996)
Metolachlor	1.9	NT (2) – M NT – Sil. <sup>9</sup> CT	. <sup>9</sup> 210	9.1	5.9 2	9-0	31.7-50.8	S	31–50, 35–38 31–56, 35–48 16–24, 34–47	$\begin{array}{c} 1.2-9.0,\\ 0.9-4.3\\ 1.4-8.0,\\ 0.8-1.8\\ 0.4-1.5,\\ 1.1-2.7\end{array}$	(Myers et al., 1995)
Metolachlor	2.2	NT CT	I	12.0	6.7 3	- S-		1	-, 0.1, -0.1, -0.01, 0.02 -0.02 -0.03, 0.03, 0.03, 0.03, 0.04	-, 0.06, <0.01, <0.01, 0.16, 0.25, 0.01, <0.01	(Hall et al., 1991)
Metolachlor	2.2	NT MT mulch CT	I	I	۱ س	-5- -5	585, 748, 486, 562	2 N	1 1 1	$\begin{array}{c} 0.03, - \\ -, 0.22 \\ 0.29, 0.55 \end{array}$	(Watts and Hall, 1996)
Metolachlor	2.24	NT CT	67 59	9.7 4.4	5.64 2 5.91			I	1 1	0.5 1.2	(Heatwole et al., 1997)
Metolachlor	1.12–1.87	TM	1 1	1 1			941, 1158, 824, 926, 1029, 860	z	$\begin{array}{c} 5,6,7,18,\\ -,1\\ 1,4,7,18,\\ 18,1\end{array}$	$\begin{array}{c} 5.5,1.0,0.9,\\ 2.1,-,0.3\\ 2.0,0.4,1.2,\\ 1.6,5.4,0.4\end{array}$	(Ghidey et al., 2005)

Table 8 (continued)

Metolachlor	0.84	$MT_{Ch}^{Ch} + Cov.C^{11}$ $MT_{Ch}^{Ch}$ $CT + Cov.C$ $CT$	I	14.5	I	$\overline{\vee}$	995, 688	Z	8, 7 14, 9 8, 6 12, 7	0.06, 0.8 0.2, 1.0 0.3, 0.5 0.4, 1.4	(Gaynor et al., 2001)
Metolachlor	0.7 0.7 Inc. <sup>10</sup> 0.7 1.4	NT (4) CT CT NT (4)	I	13.4	6.0	1–2	I	NN + S	1 1 1 1	4.22/8.28 1.01/1.97 2.22/5.98 2.44/6.01	(Rector et al., 2003)
Metribuzin	0.38	$\begin{array}{c} \mathrm{NT} \ 1\\ \mathrm{NT} \ 2\\ \mathrm{MT}_{Ch} \ 1\\ \mathrm{MT}_{Ch} \ 2\end{array}$				11 10 7	629–1012 686–1025 609–981 663–1007	Z	1-10 2-14 <0.01-1 1-15	<0.01-0.79 0.08-0.22 <0.01-0.02 <0.01-0.01	(Shipitalo et al., 1997)
Metribuzin <sup>7</sup>	0.38	$\substack{NT\\MT_{Ch}}{MT_{Dk}}$	1 1 1			10–11 7–13 6–9	976 953 960	Z	8 7 11	0.83 0.04 0.55	(Shipitalo and Owens, 2006)
Metribuzin	0.25	$\begin{split} & \mathrm{MT}_{Ch} + \mathrm{Cov.C}^{11} \\ & \mathrm{MT}_{Ch} \\ & \mathrm{CT} + \mathrm{Cov.C} \\ & \mathrm{CT} \end{split}$	I	14.5	I	$\overline{\vee}$	995, 688	Z	8, 7 14, 9 8, 6 12, 7	0.04, 1.1 0.3, 1.4 0.3, 0.6 0.3, 1.7	(Gaynor et al., 2001)
Metribuzin	0.4	$\begin{array}{l} \mathrm{NT} - \mathrm{S/W}^{13}\\ \mathrm{NT} - \mathrm{Soy}^{13}\\ \mathrm{CT} - \mathrm{S/W}\\ \mathrm{CT} - \mathrm{Soy} \end{array}$	I	18.6	6.4	ŝ	213, 191, 188	N + S	8, 7, 9 -, 14, 18 9, 9, 15 9, 5, 10	$\begin{array}{c} 0.7,0.5,0.7\\ -,1.9,0.6\\ 1.7,1.6,0.9\\ 0.9,0.8,1.4\end{array}$	(Webster and Shaw, 1996)
Metribuzin	0.816	NT CT	120	13.0	I	10	294	Z	0.2 15	0.01 0.64	(Malone et al., 1996)
Nicosulfuron	0.014	NT CT	80	5.3	5.9	0-1	I	I	1 1	2.3	(Afyuni et al., 1997)
Norflurazon	1.7	RT RT	I	18.6	6.4	e	376, 480	N+S	<i>5</i> 6	1.1 4.2,4.2 4.2,2.5 4.8 3.6	(Baughman et al., 2001)
Pendimethalin	1.0	S S S S C S S S S S S S S S S S S S S S	32	5.1	6.5	3-4	63.8 64.3 62.7 60.0	${f S-I_{tar}^{12}}$ ${f S-I_{tar}^{12}}$ ${f S-I_{cons}^{12}}$ ${f S-I_{var}}$	6 20 23 45 50	4.0, 5.0 0.4 5.0 4.1	(Potter et al., 2006)
											(continued)

Table 8 (continued)											
			Soil pr	operties					Runoff		
Pesticide	Dose kg a.i. ha <sup>-1</sup>	Tillage <sup>1</sup>	Clay	Organic carbon (OC) g kg <sup>-1</sup>	Hq	Slope %	Rainfalls mm	Type	water % applied dose	pesticide % applied dose	Reference
Simazine	1.7	NT	1	12.0	6.7	3-5	1	1	-, 0.1, < 0.01, 0.02	-, 0.18, < 0.01,	(Hall et al., 1991)
		CT				3-4			0.3, 0.3, 0.03, 0.03, 0.04	0.62, 0.51, 0.01	
Simazine	1.7	NT MT mulch CT	I	Ι	I	3-5	I	I		0.02, - -, $0.13$ 0.11, 0.35	(Watts and Hall, 1996)
Terbufos + Terbufos metabolites	1.12	NT C	I	12.8	Ι	8	63	S	9	0.03	(Felsot et al., 1990)
		NT U/D				8	64		23	0.3	
		STC				6	64		4	0.03	
		ST U/D				6	63		10	0.2	
		RT C				10	64		24	0.4	
		RT U/D				10	63		29	2.3	
		$\mathrm{MT}_{Ch}\mathrm{C}$				8	64		33	1.4	
		$MT_{Ch}$ U/D				8	64		32	1.3	
		CT C CT III				10	63 64		33 43	1.6 7 \$	
						10	5		<b>C+</b>	(··/	
Thidiazuron	0.05	ST (1) <sup>8</sup> CT	32	5.1	6.5	3-4	I	S	23 23	13.7 12.2	(Potter et al., 2003)
Tribufos	0.30	ST (1) CT	32	5.1	6.5	3-4	I	S	23 23	12.8 14.5	(Potter et al., 2003)
2,4-D	0.56	NT CT	232	21.5		10 11	100	S	0.2 3	0.02 0.27	(Kenimer et al., 1987)
<sup>1</sup> Codes of tillage prac <sup>2</sup> Hyphens are used to <sup>3</sup> Commas are used to <sup>4</sup> N: natural rainfall; S.	tices are report indicate the ra separate differ simulated rain	ted in Table 1. nge of variatio ent years of st nfall.	on of va tudy.	lues.							

<sup>6</sup> Loc: seed row localisation of the pesticide treatment. <sup>5</sup> C: contoured slope; U/D: up-and-down slope.

<sup>7</sup> Average results for 9 years of study. <sup>8</sup> Numbers in parenthesis indicate the age in years of the soil tillage. <sup>9</sup> M.: maize = residues restituted; Sil.: silage = no crop residues.

<sup>10</sup> Inc.: preplant soil incorporation of herbicide.

<sup>11</sup> Cov.C: cover crop. <sup>12</sup> I<sub>Var</sub>: variable rainfall intensity; I<sub>cons</sub>: constant rainfall intensity. <sup>13</sup> Soy: continuous soybean; S/W: soybean/wheat rotation.

most pesticide loss by runoff occurred during the first few runoff events after application (Seifert et al., 2001a; Shipitalo and Owens, 2006). In a 9-year-study comparing three types of conservation tillage (NT,  $MT_{Ch}$  and  $MT_{Dk}$ ), Shipitalo and Owens (2006) indicated that 60 to 99% of herbicide (alachlor, atrazine, linuron, and metribuzin) losses were due to the five largest transport events. After a rainfall representing 3% of annual rainfall and occurring 2 days after treatment, atrazine losses reached 4.7% of the applied dose (Shipitalo and Owens, 2003). Triplett et al. (1978) proposed a linear relationship to predict atrazine concentrations in runoff ([atrazine]<sub>runoff</sub>) based on the number of days after treatment (ln ( $[atrazine]_{runoff} =$ a + b. In (number of days after treatment)). This relationship has been tested and used for several pesticides such as alachlor (Shipitalo et al., 1997), atrazine and its metabolites (deethylatrazine and deisopropylatrazine) (Gaynor et al., 1995; Shipitalo and Owens, 2003; Shipitalo et al., 1997), cyanazine (Franti et al., 1998), linuron (Shipitalo et al., 1997), dimethipin (Potter et al., 2003), metolachlor (Gaynor et al., 1995; Ghidey et al., 2005; Webster and Shaw, 1996), metribuzin (Shipitalo et al., 1997), thidiazuron and tribufos (Potter et al., 2003). However, Hansen et al. (2001) indicated that the relationship between ln [pesticide]<sub>runoff</sub> and ln (number of days after treatment) was sensitive to annual variability in weather and may not be a good variable for simple estimates of herbicide concentration in runoff. They recommended the use of cumulative rainfall as a variable for estimating concentration in runoff:  $\ln [\text{pesticide}]_{runoff} = a + b.\ln b$ (cumulative rainfall). The intensity of the first rainfall also influences the fate of the applied molecules. A small, low-intensity rainfall occurring a few days after treatment allows the incorporation of pesticide within the topsoil horizon, thus reducing losses during subsequent runoff events (Afyuni et al., 1997; Gaynor et al., 1995; Olson et al., 1998).

As for leaching, the initial soil water content is also a key factor controlling runoff. Runoff due to exceedance of infiltration capacity was found to occur earlier under no-tillage than under conventional tillage (Isensee and Sadeghi, 1993; Rector et al., 2003; Sadeghi and Isensee, 2001). In hydromorphic soils, saturation occurred more rapidly under no-tillage and runoff was more significant than under conventional tillage (Ghidey et al., 2005). Soil tillage, even superficial, increases soil infiltration capacity and drainage, which can limit runoff (Olson et al., 1998). The time interval between two rainfalls may significantly affect runoff volumes under no-tillage. Isensee and Sadeghi (1993) indicated that, for a time interval lower than 7 days between two rainfalls, the high soil water content remaining under no-tillage, favoured by the mulch, led to a significantly higher runoff than under conventional tillage.

The mulch increases soil surface roughness, thus reducing runoff (Isensee and Sadeghi, 1993; Selim et al., 2003). However, the mulch effectiveness in controlling runoff and erosion depends on the nature and quantity of plant residues. For example, Olson et al. (1998) observed that after a soybean crop, the abundance of residues under no-tillage was not sufficient to limit (water and atrazine) runoff and higher runoff volumes than under mulch tillage were measured. For quantities of residues increasing from 0 to 1.5 t ha<sup>-1</sup> either remaining on the soil surface in no-tillage or mixed in the ploughed horizon in conventional tillage, runoff was reduced by 96 and 40%, respectively (Kenimer et al., 1987).

Strip tillage is a conservation tillage well-developed in the US which is starting to be used for spring productions by farmers in Europe. On one hand, strip tillage was found to reduce runoff compared with conventional tillage but, on the other hand, to increase lateral subsurface flow (Bosch et al., 2005). Potter et al. (2003) indicated that, after only one year of strip tillage, no effect on runoff volumes was measured compared with conventional tillage but erosion rates were lower. On the same study site, a few years later, runoff volumes were decreased by a factor of 4 compared with conventional tillage (Potter et al., 2004). Concerning pesticide losses, fluometuron losses were 2 to 3 times higher and pendimethalin losses were 12 times lower under strip tillage than under conventional tillage (Potter et al., 2006). Under strip tillage, herbicide treatments are applied only on the seed row representing only 1/3 of the field surface. In inter-row positions, the soil surface is most often covered by a cover crop, which reduces the risk of runoff (Hansen et al., 2001). Ridge tillage can also be used to reduce runoff but, in some cases, pesticides could migrate from the ridge tops to the furrows and then be transferred by runoff (Gaynor et al., 1987; Olson et al., 1998). Very few studies reported results on the effects of sub-soiling on runoff. Seifert et al. (2001a) did not observe any effect of this tillage

practice on runoff and pesticide loss compared with conventional tillage. Pesticide incorporation into soil through superficial tillage was found to significantly reduce losses by runoff (Franti et al., 1998; Olson et al., 1998; Rector et al., 2003) but this technique is not suitable for no-tillage systems and may lead to an increase in losses by leaching. Finally, whatever the tillage system, soil tillage along isotopographic lines rather than up-and-down slope was efficient in reducing transfers by runoff (Felsot et al., 1990).

# 4.4 Summary and Recommendations

Concerning pesticide transfer, it clearly appears that initial soil conditions (water content and temperature) and climatic conditions (rainfall intensity, interval between treatment and the first rainfall) play a large role in the dynamics of water and solutes and may explain the contrasted effects of tillage on pesticide transfers. Improving the understanding of pesticide transport related to tillage operations implies clarifying the temporal and spatial dynamics of solute flows, especially preferential flows, which are in this case particularly relevant. Most of the studies mentioned in this review highlighted the transient and local nature of transport mechanisms, making them difficult to characterise. In the same order of ideas, the lack of efficient and robust ways to quantify soil structure and its dynamics is an impediment to the prediction of pesticide fate and transport, especially in topsoil layers.

The significant effect of rainfall characteristics on pesticide transfers suggests that tillage effects are highly dependent on the type of climate, and its eventual modifications to come. Regional studies should be of great help in developing generic models for more robust predictions. Tillage is certainly a relevant means of controlling agricultural pesticide impacts on the environment. Pesticide fate in soil implies numerous processes in complex interactions. If one is to improve tillage systems to mitigate these impacts, one needs first to understand how the various tillage practices modify the functional characteristics of soil (water retention, hydraulic conductivity, solute transport, etc.) through detailed, comprehensive studies. Such studies should consider the diversity of tillage systems, which are currently insufficiently explored and documented. Such research efforts would imply long-term studies (>10 years) to allow significant and well-established differentiation between the cropping systems being compared and would thus involve stability in funding research programmes.

# **5** Conclusion

This review article outlines the four main following points.

- Pesticide interception is enhanced under conservation tillage practices. The intensity of this process depends on (a) the amount and nature of crop residues which have sorption capacities 10 to 60 times higher than soil, and (b) the climatic conditions, such as the timing between pesticide treatment and the first rainfall and its intensity.
- Pesticide retention, which is generally positively correlated with organic carbon content, is increased in the topsoil layer under conservation tillage. Desorption of pesticides is also affected by tillage and greater proportions of pesticides remained sorbed on soil particles and on mulch under conservation tillage.
- 3. As a consequence of points 1 and 2, a lower fraction of pesticide remains available for biological degradation under conservation tillage and, in several cases, pesticide persistence in soils increases. In addition to this lower bio-availability due to retention processes, the presence of crop residues, a lower temperature and a higher acidity under conservation tillage can disrupt or slow down microorganisms' activity. Moreover, a greater availability of carbon under conservation tillage can defer the use of pesticides as a source of carbon.
- 4. Transport of pesticides is affected by tillage management and by its interactions with climatic conditions more particularly by the intensity of rainfall, its arrival time after treatment, the interval between two rainfall events and pesticide properties e.g. water solubility, retention, half-life and formulation. In a general way, conservation tillage is more efficient in reducing runoff than leaching. Indeed, a higher soil surface roughness due to the presence of crop residues and a greater aggregate stability lead to a significant decrease in pesticide

loss in water runoff or bound to eroded sediment, while reducing tillage intensity maintains a wellconnected macropore network through which leaching of pesticides is enhanced.

## References

- Afyuni M.M., Wagger M.G., Leidy R.B. (1997) Runoff of two sulfonylurea herbicides in relation to tillage system and rainfall intensity, J. Environ. Qual. 26, 1318–1326.
- Agreste (2004) Enquête sur les pratiques culturales en 2001, Chiffres et Données - Série Agriculture - n° 159.
- Agreste (2008) Enquête sur les pratiques culturales en 2006, Chiffres et Données - Série Agriculture n° 200.
- Alletto L., Benoit P., Bergheaud V., Coquet Y. (2008) Temperature and water pressure head effects on the degradation of the diketonitrile metabolite of isoxaflutole in a loamy soil under two tillage systems, Environ. Pollut. 156, 678–688.
- Andreini M.S., Steenhuis T.S. (1990) Preferential paths of flow under conventional and conservation tillage, Geoderma 46, 85–102.
- Arshad M.A., Franzluebbers A.J., Azooz R.H. (1999) Components of surface soil structure under conventional and notillage in northwestern Canada, Soil Till. Res. 53, 41–47.
- Aubertot J.N., Barbier J.M., Carpentier A., Gril J.J., Guichard L., Lucas P., Savary S., Savini I., Voltz M. (2005) Pesticides, agriculture et environnement. Réduire l'utilisation des pesticides et en limiter les impacts environnementaux, Rapport d'Expertise scientifique collective, INRA et Cemagref (France).
- Baker J.L., Johnson H.P. (1979) The effect of tillage systems on pesticides runoff from small watersheds, T. ASAE 22, 554–559.
- Balesdent J., Mariotti A., Boisgontier D. (1990) Effect of tillage on soil organic carbon mineralization estimated from <sup>13</sup>C abundance in maize fields, J. Soil Sci. 41, 587–596.
- Banks P.A., Robinson E.L. (1982) The influence of straw mulch on the soil reception and persistence of metribuzin, Weed Sci. 30, 164–168.
- Banks P.A., Robinson E.L. (1986) Soil reception and activity of acetochlor, alachlor, and metolachlor as affected by wheat (Triticum aestivum) straw and irrigation, Weed Sci. 34, 607–611.
- Barnes C.J., Goetz A.J., Lavy T.L. (1989) Effects of imazaquin residues on cotton (*Gossypium hirsutum*), Weed Sci. 37, 820–824.
- Barr D.P., Aust S.D. (1994) Mechanisms white rot fungi use to degrade pollutants, Environ. Sci. Technol. 28, 78A.
- Barriuso E., Calvet R. (1992) Soil type and herbicide adsorption, Int. J. Environ. An. Ch. 46, 117–128.
- Barriuso E., Feller C., Calvet R., Cerri C. (1992) Sorption of atrazine, terbutryn and 2,4-D herbicides in two Brazilian Oxisols, Geoderma 53, 155–167.
- Basta N.T., Huhnke J.H., Stiegler J.H. (1997) Atrazine runoff from conservation tillage systems: a simulated rainfall study, J. Soil Water Conserv. 52, 44–48.

- Baughman T.A., Shaw D.R., Rhodes G.N., Mueller T.C. (1996) Effects of tillage on chlorimuron persistence, Weed Sci. 44, 162–165.
- Baughman T.A., Shaw D.R., Webster E.P., Boyette M. (2001) Effect of cotton (Gossypium hirsutum) tillage systems on offsite movement of fluometuron, norflurazon, and sediment in runoff, Weed Technol. 15, 184–189.
- Baumhardt R.L., Lascano R.J. (1996) Rain infiltration as affected by wheat residue amount and distribution in ridged tillage, Soil Sci. Soc. Am. J. 60, 1908–1913.
- Bedos C., Cellier P., Calvet R., Barriuso E., Gabrielle B. (2002) Mass transfer of pesticides into the atmosphere by volatilization from soils and plants: overview, Agronomie 22, 21–33.
- Berglof T., Koskinen W.C., Brucher J., Kylin H. (2000) Linuron sorption-desorption in field-moist soils, J. Agr. Food Chem. 48, 3718–3721.
- Bernoux M., Cerri C.C., Cerri C.E.P., Neto M.S., Metay A., Perrin A.S., Scopel E., Razafimbelo T., Blavet D., Piccolo M.D., Pavei M., Milne E. (2006) Cropping systems, carbon sequestration and erosion in Brazil, a review, Agron. Sustain. Dev. 26, 1–8.
- Biederbeck V.O., Campbell C.A., Hunter J.H. (1997) Tillage effects on soil microbial and biochemical characteristics in a fallow-wheat rotation in a dark brow soil, Can. J. Soil Sci. 77, 309–316.
- Blevins R.L., Frye W.W. (1993) Conservation tillage: an ecological approach to soil management, Adv. Agron. 51, 33–78.
- Bosch D.D., Potter T.L., Truman C.C., Bednarz C.W., Strickland T.C. (2005) Surface runoff and lateral subsurface flow as a response to conservation tillage and soil-water conditions, T. ASAE 48, 2137–2144.
- Bouché M.B. (1972) Lombriciens de France. Écologie et systématique, Annales de zoologie-écologie animale, numéro hors-série, Ed. INRA, 671 p.
- Boyd S.A., Xiangcan J., Lee J.F. (1990) Sorption of nonionic organic compounds by corn residues from a no-tillage field, J. Environ. Qual. 19, 734–738.
- Bragagnolo N., Mielniczuk J. (1990) Mulch from eight crop sequences and its effects on soil temperature, moisture, emergence and initial growth of corn, Rev. Bras. Cienc. Solo 14, 91–98.
- Brown B.A., Hayes R.M., Tyler D.D., Mueller T.C. (1994) Effect of tillage and cover crop on fluometuron adsorption and degradation under controlled conditions, Weed Sci. 42, 629–634.
- Brown B.A., Hayes R.M., Tyler D.D., Mueller T.C. (1996) Effect of long-term Vetch (*Vicia villosa*) cover crop and tillage system on fluometuron dissipation from surface soil, Weed Sci. 44, 171–175.
- Chapman R.A., Cole C.M. (1982) Observations on the influence of water and soil pH on the persistance of insecticides, J. Environ. Sci. Health 17, 487–504.
- Chin Y.P., Weber W.J., Eadie B.L. (1990) Estimating the effects of dispersed organic polymers on the sorption of contaminants by natural solids. 2. Sorption in the presence of humic and other natural macromolecules, Environ. Sci. Technol. 24, 837–842.
- Clay S.A., Koskinen W.C. (1990) Adsorption and desorption of atrazine, hydroxyatrazine, and s-glutathione atrazine on two soils, Weed Sci. 38, 262–266.

- Clay S.A., Koskinen W.C., Carlson P. (1991) Alachlor movement through intact soil columns taken from two tillage systems, Weed Technol. 5, 485–489.
- Clay S.A., Clay D.E., Koskinen W.C., Berg R.K. (1998) Application method: impacts on atrazine and alachlor movement, weed control, and corn yield in three tillage systems, Soil Till. Res. 48, 215–224.
- Coats K.H., Smith B.D. (1964) Dead-end pore volume and dispersion in porous media, SPE J. 4, 73–84.
- Comia R.A., Stenberg M., Nelson P., Rydberg T., Håkansson I. (1994) Soil and crop responses to different tillage systems, Soil Till. Res. 29, 335–355.
- CTIC (2006) National crop residues management survey, Conservation tillage information center, Ames, IA.
- Curran W.S., Liebl R.A., Simmons F.W. (1992) Effects of tillage and application method on clomazone, imazaquin, and imazethapyr persistence, Weed Sci. 40, 482–489.
- Dalal R.C., Henderson P.A., Glasby J.M. (1991) Organic matter and microbial biomass in a vertisol after 20 yr of zero-tillage, Soil Biol. Biochem. 23, 435–441.
- Dao T.H. (1991) Field decay of wheat straw and its effects on metribuzin and S-ethyl metribuzin sorption and elution from crop residues, J. Environ. Qual. 20, 203–208.
- Dao T.H., Lavy T.L. (1978) Atrazine adsorption on soil as influence by temperature, moisture content and electrolyte concentration, Weed Sci. 26, 303–309.
- Ding G., Novak J.M., Amarasiriwardena D., Hunt P.G., Xing B. (2002a) Soil organic matter characteristics as affected by tillage management, Soil Sci. Soc. Am. J. 66, 421–429.
- Ding G.W., Novak J.M., Herbert S., Xing B.S. (2002b) Longterm tillage effects on soil metolachlor sorption and desorption behavior, Chemosphere 48, 897–904.
- Ding G.W., Liu X.B., Herbert S., Novak J., Amarasiriwardena D., Xing B.H. (2006) Effect of cover crop management on soil organic matter, Geoderma 130, 229–239.
- Doran J.W. (1980) Soil microbial and biochemical changes associated with reduced tillage, Soil Sci. Soc. Am. J. 44, 765–771.
- Drijber R.A., Doran J.W., Parkhurst A.M., Lyon D.J. (2000) Changes in soil microbial community structure with tillage under long-term wheat-fallow management, Soil Biol. Biochem. 32, 1419–1430.
- Drury C.F., Tan C.S., Welacky T.W., Oloya T.O., Hamill A.S., Weaver S.E. (1999) Red clover and tillage influence on soil temperature, water content, and corn emergence, Agron. J. 91, 101–108.
- Düring R.A., Hummel H.E. (1993) Soil tillage as a parameter influencing the fate of three selected soil herbicides, Med. Fac. Landbouww. Univ. Gent 58/3a, 827–835.
- Düring R.A., Hummel H.E. (1999) Herbicide and metabolite movement in different soils as studied by computer assisted microlysimeters, Chemosphere 39, 641–654.
- Düring R.A., Hoss T., Gath S. (2002) Depth distribution and bioavailability of pollutants in long-term differently tilled soils, Soil Till. Res. 66, 183–195.
- Edwards W.M., Norton L.D., Redmond C.E. (1988) Characterizing macropores that affect infiltration in nontilled soil, Soil Sci. Soc. Am. J. 52, 483–487.
- Edwards W.M., Shipitalo M.J., Owens L.B., Norton L.D. (1989) Water and nitrate movement in eathworm burrows within

long-term no-till cornfields, J. Soil Water Conserv. 44, 240-243.

- Edwards W.M., Shipitalo M.J., Dick W.A., Owens L.B. (1992a) Rainfall intensity affects transport of water and chemicals through macropores in no-till soil, Soil Sci. Soc. Am. J. 56, 52–58.
- Edwards W.M., Shipitalo M.J., Traina S.J., Edwards C.A., Owens L.B. (1992b) Role of *Lumbricus terrestris* (L.) burrows on quality of infiltrating water, Soil Biol. Biochem. 24, 1555–1561.
- Ela S.D., Gupta S.C., Rawles W.J. (1992) Macropore and surface seal interactions affecting water infiltration into soil, Soil Sci. Soc. Am. J. 56, 714–721.
- Erbach D.C., Lovely W.G. (1975) Effect of plant residue on herbicide performance in no-tillage corn, Weed Sci. 23, 512–515.
- Essington M.E., Tyler D.D., Wilson G.V. (1995) Fluometuron behavior in long-term tillage plots, Soil Sci. 160, 405–414.
- Fawcett R.S., Christensen B.R., Tierney D.P. (1994) The impact of conservation tillage on pesticide runoff into surface water: a review and analysis, J. Soil Water Conserv. 49, 126–135.
- Felsot A.S., Mitchell J.K., Kenimer A.L. (1990) Assessment of management practices for reducing pesticide runoff from sloping cropland in Illinois, J. Environ. Qual. 19, 539–545.
- Ferri M.V.W., Vidal R.A., Gomes J., Dick D.P., de Souza R.F. (2002) Activity of the herbicide acetochlor in soil under notill and conventional tillage system, Pesqui. Agropecu. Bras. 37, 1697–1703.
- Flury M., Leuenberger J., Studer B., Flühler H. (1995) Transport of anions and herbicides in a loamy and a sandy field soil, Water Resour. Res. 30, 1945–1954.
- Fomsgaard I.S., Spliid N.H., Felding G. (2003) Leaching of pesticides through normal-tillage and low-tillage soil - A lysimeter study. II, Glyphosate J. Environ. Sci. Health B. 38, 19–35.
- Fortin J., Gagnon-Bertrand E., Vezina L., Rompre M. (2002) Preferential bromide and pesticide movement to tile drains under different cropping practices, J. Environ. Qual. 31, 1940–1952.
- Franti T.G., Peter C.J., Tierney D.P., Fawcett R.S., Myers J.L. (1998) Reducing herbicide losses from tile-outlet terraces, J. Soil Water Conserv. 53, 25–31.
- Gan J., Becker R.L., Koskinen W.C., Buhler D.D. (1996) Degradation of atrazine in two soils as a function of concentration, J. Environ. Qual. 25, 1064–1072.
- Gao J.P., Maguhn J., Spitzauer P., Kettrup A. (1998) Sorption of pesticides in the sediment of the Teufelsweiher pond (Southern Germany). II: Competitive adsorption, desorption of aged residues and effect of dissolved organic carbon, Water Res. 32, 2089–2094.
- Gaston L.A., Locke M.A. (1996) Bentazon mobility through intact, unsaturated columns of conventional and no-till Dundee soil, J. Environ. Qual. 25, 1350–1356.
- Gaston L.A., Locke M.A. (2000) Acifluorfen sorption, degradation, and mobility in a Mississippi delta soil, Soil Sci. Soc. Am. J. 64, 112–121.
- Gaston L.A., Locke M.A., Zablotowicz R.M. (1996) Sorption and degradation of bentazon in conventional- and no-till Dundee soil, J. Environ. Qual. 25, 120–126.
- Gaston L.A., Boquet D.J., Bosch M.A. (2001) Fluometuron wash-off from cover crop residues and fate in a loessial soil, Soil Sci. 166, 681–690.

- Gaston L.A., Boquet D.J., Bosch M.A. (2003) Fluometuron sorption and degradation in cores of silt loam soil from different tillage and cover crop systems, Soil Sci. Soc. Am. J. 67, 747–755.
- Gaynor J.D., Stone J.A., Vyn T.J. (1987) Tillage systems and atrazine and alachlor residues on a poorly drained soil, Can. J. Soil Sci. 67, 959–963.
- Gaynor J.D., MacTavish D.C., Findlay W.I. (1995) Atrazine and metolachlor loss in surface and subsurface runoff from three tillage treatments in corn, J. Environ. Qual. 24, 246–256.
- Gaynor J.D., MacTavish D.C., Labaj A.B. (1998) Atrazine and metolachlor residues in Brookston CL following conventional and conservation tillage culture, Chemosphere 36, 3199–3210.
- Gaynor J.D., Tan C.S., Ng H.Y.F., Drury C.F., Welacky T.W., vanWesenbeeck I.J. (2000) Tillage and controlled drainagesubirrigated management effects on soil persistence of atrazine, metolachlor, and metribuzin in corn, J. Environ. Qual. 29, 936–947.
- Gaynor J.D., Tan C.S., Drury C.F., Ng H.Y.F., Welacky T.W., van Wesenbeeck I.J. (2001) Tillage, intercrop, and controlled drainage-subirrigation influence atrazine, metribuzin, and metolachlor loss, J. Environ. Qual. 30, 561–572.
- Gebhardt M.R., Daniel T.C., Schweizer E.E., Allmaras R.R. (1985) Conservation tillage, Science 230, 625–630.
- Ghadiri H., Shea P.J., Wicks G.A., Haderlie L.C. (1984) Atrazine dissipation in conventional-till and no-till sorghum, J. Environ. Qual. 13, 549–552.
- Ghidey F., Blanchard P.E., Lerch R.N., Kitchen N.R., Alberts E.E., Sadler E.J. (2005) Measurement and simulation of herbicide transport from the corn phase of three cropping systems, J. Soil Water Conserv. 60, 260–273.
- Gish T.J., Helling C.S., Mojasevic M. (1991) Preferential movement of atrazine and cyanazine under field conditions, T. ASAE 34, 1699–1705.
- Gish T.J., Shirmohammadi A., Wienhold B.J. (1994) Fieldscale mobility and persistence of commercial and starchencapsulated atrazine and alachlor, J. Environ. Qual. 23, 355–359.
- Gish T.J., Shirmohammadi A., Vyravipillai R., Wienhold B.J. (1995) Herbicide leaching under tilled and no-tillage fields, Soil Sci. Soc. Am. J. 59, 895–901.
- Glotfelty D.E. (1987) The effects of conservation tillage practices on pesticide volatilization and degradation, in: Logan T.J. et al. (Eds.), Effects of conservation tillage on groundwater quality: Nitrates and pesticides. Lewis, Chelsea, MI, pp. 56–62.
- Granovsky A.V., McCoy E.L., Dick W.A., Shipitalo M.J., Edwards W.M. (1993) Water and chemical transport through long-term no-till and plowed soils, Soil Sci. Soc. Am. J. 57, 1560–1567.
- Grant R.F., Izaurralde R.C., Chanasyk D.S. (1990) Soil temperature under conventional and minimum tillage: simulation and experimental verification, Can. J. Soil Sci. 70, 289–304.
- Grey T.L., Walker R.H., Wehtje G.R.H., Hancock H.G. (1997) Sulfentrazone adsorption and mobility as affected by soil and pH, Weed Sci. 45, 733–738.
- Gupta S.C., Schneider E.C., Swan J.B. (1988) Planting depth and tillage interactions on corn emergence, Soil Sci. Soc. Am. J. 52, 1122–1127.

- Hall J.K., Mumma R.O. (1994) Dicamba mobility in conventionally tilled and non-tilled soil, Soil Till. Res. 30, 3–17.
- Hall J.K., Murray M.R., Hartwig N.L. (1989) Herbicide leaching and distribution in tilled and untilled soil, J. Environ. Qual. 18, 439–445.
- Hall J.K., Mumma R.O., Watts D.W. (1991) Leaching and runoff losses of herbicides in a tilled and untilled field, Agr. Ecosyst. Environ. 37, 303–314.
- Hall J.K., Jones G.A., Hickman M.V., Amistadi M.K., Bogus E.R., Mumma R.O., Hartwig N.L., Hoffman L.D. (1998) Formulation and adjuvant effects on leaching of atrazine and metolachlor, J. Environ. Qual. 27, 1334–1347.
- Hansen N.C., Moncrief J.F., Gupta S.C., Capel P.D., Olness A.E. (2001) Herbicide banding and tillage system interactions on runoff losses of alachlor and cyanazine, J. Environ. Qual. 30, 2120–2126.
- Hanson J. (2006) History of tillage. Improving Soil Enhancing Profit. Manitoba-North Dakota Zero Tillage Farmers Association Workshop, Bismarck, ND, 9–10 February 2006.
- Harman W.L., Wang E., Williams J.R. (2004) Reducing atrazine losses: Water quality implications of alternative runoff control practices, J. Environ. Qual. 33, 7–12.
- Heatwole C.D., Zacharias S., Mostaghimi S., Dillaha T.A. (1997) Movement of field-applied atrazine, metolachlor, and bromide in a sandy loam soil, T. ASAE 40, 1267–1276.
- Isensee A.R., Sadeghi A.M. (1993) Impact of tillage practice on runoff and pesticide transport, J. Soil Water Conserv. 48, 523–527.
- Isensee A.R., Sadeghi A.M. (1994) Effects of tillage and rainfall on atrazine residue levels in soil, Weed Sci. 42, 462–467.
- Isensee A.R., Sadeghi A.M. (1996) Effect of tillage reversal on herbicide leaching to groundwater, Soil Sci. 161, 382–389.
- Isensee A.R., Sadeghi A.M. (1997) Interactions of tillage and rainfall on atrazine leaching under field and laboratory conditions, Chemosphere 34, 2715–2723.
- Isensee A.R., Nash R.G., Helling C.S. (1990) Effect of conventional vs. no-tillage on pesticide leaching to shallow groundwater, J. Environ. Qual. 19, 434–440.
- Jaynes D.B., Ahmed S.I., Kung K.J.S., Kanwar R.S. (2001) Temporal dynamics of preferential flow to a subsurface drain, Soil Sci. Soc. Am. J. 65, 1368–1376.
- Johnson M.D., Wyse D.L., Lueschen W.E. (1989) The influence of herbicide formulation on weed control in four tillage systems, Weed Sci. 37, 239–249.
- Kabir Z. (2005) Tillage or no-tillage: Impact on mycorrhizae, Can. J. Plant Sci. 85, 23–29.
- Kasteel R., Garnier P., Vachier P., Coquet Y. (2007) Dye tracer infiltration in the plough layer after straw incorporation, Geoderma 137, 360–369.
- Kenimer A.L., Mostaghimi S., Young R.W., Dillaha T.A., Shanholtz V.O. (1987) Effects of residue cover on pesticide losses from conventional and no-tillage systems, T. ASAE 30, 953–959.
- Kitchen N.R., Hughes D.E., Donald W.W., Alberts E.E. (1998) Agrichemical movement in the root-zone of claypan soils: ridge- and mulch-tillage systems compared, Soil Till. Res. 48, 179–193.
- Kung K.J.S., Steenhuis T.S., Kladivko E.J., Gish T.J., Bubenzer G., Helling C.S. (2000) Impact of preferential flow on the transport of adsorbing and non-adsorbing tracers, Soil Sci. Soc. Am. J. 64, 1290–1296.

- Lafrance P., Marineau L., Perreault L., Villeneuve J.P. (1994) Effect of natural dissolved organic matter found in groundwater on soil adsorption and transport of pentachlorophenol, Environ. Sci. Technol. 28, 2314–2320.
- Lal R., Mahboubi A.A., Fausey N.R. (1994) Long-term tillage and rotation effects on properties of a Central Ohio Soil, Soil Sci. Soc. Am. J. 58, 517–522.
- Lal R., Reicosky D.L., Hanson J.D. (2007) Evolution of the plow over 10,000 years and the rationale for no-till farming, Soil Till. Res. 93, 1–12.
- Lamb J.A., Dowdy R.H., Anderson J.L., Allmaras R.R. (1998) Water quality in an irrigated sandy soil: ridge tillage in rotated corn and soybean compared with full-width tillage in continuous corn, Soil Till. Res. 48, 167–177.
- Lavorenti A., Rocha A.A., Prata F., Regitano J.B., Tornisielo V.L., Pinto O.B. (2003) Reactions of diclosulam in a Rhodic Hapludox soil under no-till and conventional tillage systems, Rev. Bras. Cienc. Solo 27, 183–190.
- Levanon D., Codling E.E., Meisinger J.J., Starr J.L. (1993) Mobility of agrochemicals through soil from two tillage systems, J. Environ. Qual. 22, 155–161.
- Levanon D., Meisinger J.J., Codling E.E., Starr J.L. (1994) Impact of tillage on microbial activity and the fate of pesticides in the upper soil, Water Air Soil Poll. 72, 179–189.
- Li K., Liu W.P., Xu D.M., Lee S.J. (2003) Influence of organic matter and pH on bentazone sorption in soils, J. Agr. Food Chem. 51, 5362–5366.
- Li K., Xing B.S., Torello W.A. (2005) Effect of organic fertilizers derived dissolved organic matter on pesticide sorption and leaching, Environ. Pollut. 134, 187–194.
- Locke M.A. (1992) Sorption-desorption kinetics of alachlor in surface soil from two soybean tillage systems, J. Environ. Qual. 21, 558–566.
- Locke M.A., Zablotowicz R.M., Gaston L.A. (1995) Fluometuron herbicide interactions in crop residue-managed soils, in: Buehring N.W., Kingery W.A. (Eds.), Proc. Southern Conservation Tillage Conf. for Sustainable Agriculture: A focus on water quality. Jackson, MS. MAFES, Mississippi State University, Mississippi State, Starkville, MS, pp. 55–58.
- Locke M.A., Gaston L.A., Zablotowicz R.M. (1996) Alachlor biotransformation and sorption in soil from two soybean tillage systems, J. Agr. Food Chem. 44, 1128–1134.
- Locke M.A., Harper S.S. (1991a) Metribuzin degradation in soil: I- Effects of residue amendment, metribuzin level, and soil depth, Pestic. Sci. 31, 221–237.
- Locke M.A., Harper S.S. (1991b) Metribuzin degradation in soil: II- Effects of tillage, Pestic. Sci. 31, 239–247.
- Locke M.A., Bryson C.T. (1997) Herbicide-soil interactions in reduced tillage and plant residue management systems, Weed Sci. 45, 307–320.
- Locke M.A., Zablotowicz R.M., Bauer P.J., Steinriede R.W., Gaston L.A. (2005) Conservation cotton production in the southern United States: herbicide dissipation in soil and cover crops, Weed Sci. 53, 717–727.
- Logan T.J., Eckert D.J., Beak D.G. (1994) Tillage, crop and climatic effects on runoff and tile drainage losses of nitrate and four herbicides, Soil Till. Res. 30, 75–103.
- Loux M.M., Liebl R.A., Slife F.W. (1989) Availability and persistence of imazaquin, imazethapyr, and clomazone in soil, Weed Sci. 37, 259–267.

- Lowery B., Hartwig R.C., Stoltenberg D.E., Fermanich K.J., McSweeney K. (1998) Groundwater quality and crop-yield responses to tillage management on a Sparta sand, Soil Till. Res. 48, 225–237.
- Ma L.W., Selim H.M. (2005) Predicting pesticide transport in mulch-amended soils: A two-compartment model, Soil Sci. Soc. Am. J. 69, 318–327.
- Malone R.W., Warner R.C., Byers M.E. (1996) Runoff losses of surface-applied metribuzin as influenced by yard waste compost amendments, no-tillage, and conventional-tillage, B. Environ. Contam. Tox. 57, 536–543.
- Malone R.W., Shipitalo M.J., Ma L., Ahuja L.R., Rojas K.W. (2001) Macropore component assessment of the root zone water quality model (RZWQM) using no-till soil blocks, T. ASAE 44, 843–852.
- Martin C.D., Baker J.L., Erbach D.C., Johnson H.P. (1978) Washoff of herbicides applied to corn residue, T. ASAE 21, 1164–1168.
- Masse L., Patni N.K., Jui P.Y., Clegg B.S. (1996) Tile effluent quality and chemical losses under conventional and no tillage - Part 2: atrazine and metolachlor, T. ASAE 39, 1673–1679.
- Masse L., Patni N.K., Jui P.Y., Clegg B.S. (1998) Groundwater quality under conventional and no tillage: II. Atrazine, deethylatrazine, and metolachlor, J. Environ. Qual. 27, 877–883.
- Masutti C. (2004) Le Dust Bowl, la politique de conservation des ressources et les écologues aux États-Unis dans les années 1930, Thèse de Doctorat de l'Université Louis Pasteur.
- Mazzoncini M., Lorenzi R., Risaliti R., Sorce C., Ginanni M., Curadi M., Pini R. (1998) Diclofop-methyl dissipation in clay soil under different tillage systems in central Italy, Soil Till. Res. 46, 241–250.
- Mills J.A., Witt W.W. (1991) Dissipation of imazaquin and imazethapyr under conventional and no-tillage soybean (Glycine max), Weed Technol. 5, 586–591.
- Mills J.A., Witt W.W., Barrett M. (1989) Effects of tillage on the efficacy and persistence of clomazone in soybean (Glycine max), Weed Sci. 37, 233–238.
- Monks C.D., Banks P.A. (1993) Effect of straw, ash, and tillage on dissipation of imazaquin and imazethapyr, Weed Sci. 41, 133–137.
- Moonen A.C., Barberi P. (2004) Size and composition of the weed seedbank after 7 years of different cover-crop-maize management systems, Weed Res. 44, 163–177.
- Myers J.L., Wagger M.G., Leidy R.B. (1995) Chemical movement in relation to tillage system and simulated rainfall intensity, J. Environ. Qual. 24, 1183–1192.
- Nanny M.A., Maza J.P. (2001) Noncovalent interactions between monoaromatic compounds and dissolved humic acids: A deuterium NMR T-1 relaxation study, Environ. Sci. Technol. 35, 379–384.
- Novak J.M., Watts D.W., Hunt P.G. (1996) Long-term tillage effects on atrazine and fluometuron sorption in coastal plain soils, Agr. Ecosyst. Environ. 60, 165–173.
- Ochsner T.E., Stephens B.M., Koskinen W.C., Kookana R.S. (2006) Sorption of a hydrophilic pesticide: Effects of soil water content, Soil Sci. Soc. Am. J. 70, 1991–1997.
- Ogden C.B., van Es H.M., Wagenet R.J., Steenhuis T.S. (1999) Spatial-temporal variability of preferential flow in a clay soil under no-till and plow-till, J. Environ. Qual. 28, 1264–1273.

- Olson B.L.S., Regehr D.L., Janssen K.A., Barnes P.L. (1998) Tillage system effects on atrazine loss in surface water runoff, Weed Technol. 12, 646–651.
- Otto S., Riello L., During R.A., Hummel H.E., Zanin G. (1997) Herbicide dissipation and dynamics modelling in three different tillage systems, Chemosphere 34, 163–178.
- Peigné J., Ball B.C., Roger-Estrade J., David C. (2007) Is conservation tillage suitable for organic farming? A review, Soil Use Manage. 23, 129–144.
- Phillips R.E., Quisenberry V.L., Zeleznik J.M., Dunn G.H. (1989) Mechanism of water entry into simulated macropores, Soil Sci. Soc. Am. J. 53, 1629–1635.
- Pinheiro E.F.M., Pereira M.G., Anjos L.H.C. (2004) Aggregate distribution and soil organic matter under different tillage systems for vegetable crops in a Red Latosol from Brazil, Soil Till. Res. 77, 79–84.
- Pot V., Simunek J., Benoit, P., Coquet Y., Yra A., Martinez-Cordon M. J. (2005) Impact of rainfall intensity on the transport of two herbicides in undisturbed grassed filter strip soil cores, J. Contam. Hydrol. 81, 63–88.
- Potter T.L., Truman C.C., Bosch D.D., Bednarz C.W. (2003) Cotton defoliant runoff as a function of active ingredient and tillage, J. Environ. Qual. 32, 2180–2188.
- Potter T.L., Truman C.C., Bosch D.D., Bednarz C. (2004) Fluometuron and pendimethalin runoff from strip and conventionally tilled cotton in the southern Atlantic Coastal Plain, J. Environ. Qual. 33, 2122–2131.
- Potter T.L., Truman C.C., Strickland T.C., Bosch D.D., Webster T.M., Franklin D.H., Bednarz C.W. (2006) Combined effects of constant versus variable intensity simulated rainfall and reduced tillage management on cotton preemergence herbicide runoff, J. Environ. Qual. 35, 1894–1902.
- Preston C.M., Newman R.H., Rother P. (1994) Using <sup>13</sup>C CPMAS NMR to assess the effects of cultivation on the organic matter of particle size fractions in a grassland soil, Soil Sci. 157, 26–35.
- Quisenberry V.L., Phillips R.E., Zeleznik J.M. (1994) Spatial distribution of water and chloride macropore flow in a wellstructured soil, Soil Sci. Soc. Am. J. 58, 1294–1300.
- Rasmussen K.J. (1991) Reduced soil tillage and Italian ryegrass as catch crop. II. Soil bulk density, root development and soil chemistry, Dan. J. Plant Soil Sci. 95, 139–154.
- Réal B., Malaval C., Bonin L., Labreuche J., Barriuso E., Benoit P., Bedos C., Köller R., Heddadj D., Quéré L., Cariolle M., Caboulet D., Alletto L. (2007) Impacts des TCSL sur les transferts de produits phytosanitaires, Évaluation des impacts environnementaux des techniques culturales sans labour (TCSL) en France, Edition ADEME, pp. 250–268.
- Rector R.J., Regehr D.L., Barnes P.L., Loughin T.M. (2003) Atrazine, S-metolachlor, and isoxaflutole loss in runoff as affected by rainfall and management, Weed Sci. 51, 810–816.
- Reddy K.N., Locke M.A. (1996) Imazaquin spray retention, foliar washoff, and runoff losses under simulated rainfall, Pestic. Sci. 48, 179–187.
- Reddy K.N., Locke M.A. (1998) Sulfentrazone sorption, desorption, and mineralization in soils from two tillage systems, Weed Sci. 46, 494–500.
- Reddy K.N., Zablotowicz R.M., Locke M.A. (1995a) Chlorimuron adsorption, desorption, and degradation in soils from conventional tillage and no-tillage systems, J. Environ. Qual. 24, 760–767.

- Reddy K.N., Locke M.A., Wagner S.C., Zablotowicz R.M., Gaston L.A., Smeda R.J. (1995b) Chlorimuron ethyl sorption and desorption kinetics in soils and herbicide-dessicated cover crop residues, J. Agr. Food Chem. 43, 2752–2757.
- Reddy K.N., Locke M.A., Gaston L.A. (1997a) Tillage and cover crop effects on cyanazine adsorption and desorption kinetics, Soil Sci. 162, 501–509.
- Reddy K.N., Locke M.A., Zablotowicz R.M. (1997b) Soil type and tillage effects on sorption of cyanazine and degradation products, Weed Sci. 45, 727–732.
- Renner K.A., Schabenberger O., Kells J.J. (1998) Effect of tillage and application method on corn (*Zea mays*) response to imidazolinone residues in soil, Weed Technol. 12, 281–285.
- Rhoton F.E., Shipitalo M.J., Lindbo D.L. (2002) Runoff and soil loss from midwestern and southeastern US silt loam soils as affected by tillage practice and soil organic matter content, Soil Till. Res. 66, 1–11.
- Ritter W.F., Chirnside A.E.M., Scarborough R.W. (1996) Movement and degradation of triazines, alachlor, and metolachlor in sandy soils, J. Environ. Sci. Health 31, 2699–2721.
- Rovira A.D., Smettem K.R.J., Lee K.E. (1987) Effect of rotation and conservation tillage on earthworms in a Red-brown earth under wheat, Aust. J. Soil Res. 38, 829–834.
- Sadeghi A.M., Isensee A.R. (1992) Effect of tillage systems and rainfall patterns on atrazine distribution in soil, J. Environ. Qual. 21, 464–469.
- Sadeghi A.M., Isensee A.R. (1996) Impact of reversing tillage practices on movement and dissipation of atrazine in soil, Soil Sci. 161, 390–397.
- Sadeghi A.M., Isensee A.R. (1997) Alachlor and cyanazine persistence in soil under different tillage and rainfall regimes, Soil Sci. 162, 430–438.
- Sadeghi A.M., Isensee A.R. (2001) Impact of hairy vetch cover crop on herbicide transport under field and laboratory conditions, Chemosphere 44, 109–118.
- Sadeghi A.M., Isensee A.R., Shelton D.R. (1998) Effect of tillage age on herbicide dissipation: A side-by-side comparison using microplots, Soil Sci. 163, 883–890.
- Sadeghi A.M., Isensee A.R., Shirmohammadi A. (2000) Influence of soil texture and tillage on herbicide transport, Chemosphere 41, 1327–1332.
- Salloum M.J., Dudas M.J., McGill W.B. (2001) Variation of 1naphthol sorption with organic matter fractionation: the role of physical conformation, Org. Geochem. 32, 709–719.
- Satchell J.E. (1983) Earthworm Ecology from Darwin to vermiculture. Ed. Chapman and Hall, London, 495 p.
- Sauer T.J., Daniel T.C. (1987) Effect of tillage system on runoff losses of surface-applied pesticides, Soil Sci. Soc. Am. J. 51, 410–415.
- Seifert S., Shaw D.R., Kingery W.L., Snipes C.E., Wesley R.A. (2001a) Imazaquin mobility and persistence in a Sharkey clay soil as influenced by tillage systems, Weed Sci. 49, 571–577.
- Seifert S., Shaw D.R., Zablotowicz R.M., Wesley R.A., Kingery W.L. (2001b) Effect of tillage on microbial characteristics and herbicide degradation in a Sharkey clay soil, Weed Sci. 49, 685–693.
- Selim H.M., Zhou L., Zhu H. (2003) Herbicide retention in soil as affected by sugarcane mulch residue, J. Environ. Qual. 32, 1445–1454.
- Senesi N., Testini C. (1982) Physico-chemical investigations of interaction mechanisms between s-triazine herbicides and soil humic acids, Geoderma 28, 129–146.
- Seta A.K., Blevins R.L., Frye W.W., Barfield B.J. (1993) Reducing soil erosion and agricultural chemical losses with conservation tillage, J. Environ. Qual. 22, 661–665.
- Seyfried M.S., Rao P.S.C. (1987) Solute transport in undisturbed columns of an aggregated tropical soil: preferential flow effects, Soil Sci. Soc. Am. J. 51, 1434–1444.
- Shang C., Arshad M.A. (1998) Sorption of clopyralid, dicamba and MCPA by two soils with conventional and no-till management, Can. J. Soil Sci. 78, 181–186.
- Shelton D.R., Sadeghi A.M., Isensee A.R. (1998) Effect of tillage on atrazine bioavailability, Soil Sci. 163, 891–896.
- Shipitalo M.J., Edwards W.M. (1996) Effects of initial water content on macropore/matrix flow and transport of surfaceapplied chemicals, J. Environ. Qual. 25, 662–670.
- Shipitalo M.J., Owens L.B. (2003) Atrazine, deethylatrazine, and deisopropylatrazine in surface runoff from conservation tilled watersheds, Environ. Sci. Technol. 37, 944–950.
- Shipitalo M.J., Owens L.B. (2006) Tillage system, application rate, and extreme event effects on herbicide losses in surface runoff, J. Environ. Qual. 35, 2186–2194.
- Shipitalo M.J., Edwards W.M., Dick W.A., Owens L.B. (1990) Initial storm effects on macropore transport of surfaceapplied chemicals in no-till soil, Soil Sci. Soc. Am. J. 54, 1530–1536.
- Shipitalo M.J., Edwards W.M., Owens L.B. (1997) Herbicide losses in runoff from conservation-tilled watersheds in a corn-soybean rotation, Soil Sci. Soc. Am. J. 61, 267–272.
- Sigua G.C., Isensee A.R., Sadeghi A.M. (1993) Influence of rainfall intensity and crop residue on leaching of atrazine through intact no-till soil cores, Soil Sci. 156, 225–232.
- Sigua G.C., Isensee A.R., Sadeghi A.M. (1995) Influence of tillage, antecedent moisture, and rainfall timing on atrazine transport, Weed Sci. 43, 134–139.
- Singh N., Kloeppel H., Klein W. (2002) Movement of metolachlor and terbuthylazine in core and packed soil columns, Chemosphere 47, 409–415.
- Singh P., Kanwar R.S. (1991) Preferential solute transport through macropores in large undisturbed saturated soil columns, J. Environ. Qual. 20, 295–300.
- Six J., Elliott E.T., Paustian K. (1999) Aggregate and soil organic matter dynamics under conventional and no-tillage systems, Soil Sci. Soc. Am. J. 63, 1350–1358.
- Sleutel S., Kader M.A., Leinweber P., D'Haene K., De Neve S. (2007) Tillage management alters surface soil organic matter composition: A pyrolysis mass spectroscopy study, Soil Sci. Soc. Am. J. 71, 1620–1628.
- Sorenson B.A., Shea P.J., Roeth F.W. (1991) Effects of tillage, application time and rate on metribuzin dissipation, Weed Res. 31, 333–345.
- Stearman G.K., Lewis R.J., Tortorelli L.J., Tyler D.D. (1989) Herbicide reactivity of soil organic matter fractions in notilled and tilled cotton, Soil Sci. Soc. Am. J. 53, 1690–1694.
- Stehouwer R.C., Dick W.A., Traina S.J. (1993) Characteristics of earthworm burrow lining affecting atrazine sorption, J. Environ. Qual. 22, 181–185.
- Suba J.D., Essington M.E. (1999) Adsorption of fluometuron and norflurazon: Effect of tillage and dissolved organic carbon, Soil Sci. 164, 145–155.

- Szajdak L., Jezierski A., Cabrera M.L. (2003) Impact of conventional and no-tillage management on soil amino acids, stable and transient radicals and properties of humic and fulvic acids, Org. Geochem. 34, 693–700.
- Tatzber M., Stemmer M., Spiegel H., Katzlberger C., Haberhauer G., Gerzabek M. H. (2008) Impact of different tillage practices on molecular characteristics of humic acids in a long-term field experiment – An application of three different spectroscopic methods, Sci. Total Environ. 406, 256–268.
- Teasdale J.R., Shelton D.R., Sadeghi A.M., Isensee A.R. (2003) Influence of hairy vetch residue on atrazine and metolachlor soil solution concentration and weed emergence, Weed Sci. 51, 628–634.
- Tebrügge F., During R.A. (1999) Reducing tillage intensity a review of results from a long-term study in Germany, Soil Till. Res. 53, 15–28.
- Torresen K.S., Skuterud R., Tandsaether H.J., Hagemo M.B. (2003) Long-term experiments with reduced tillage in spring cereals. I. Effects on weed flora, weed seedbank and grain yield, Crop Prot. 22, 185–200.
- Triplett G.B., Conner B.J., Edwards W.M. (1978) Transport of atrazine and simazine in runoff from conventional and notillage corn, J. Environ. Qual. 7, 77–84.
- Trojan M.D., Linden D.R. (1992) Microrelief and rainfall effects on water and solute movement in earthworm burrows, Soil Sci. Soc. Am. J. 56, 727–733.
- Ulbrich A.V., Souza J.R.P., Shaner D. (2005) Persistence and carryover effect of imazapic and imazapyr in Brazilian cropping systems, Weed Technol. 19, 986–991.
- Unger P.W. (1987) Straw mulch effects on soil temperature and sorghum germination and growth, Agron. J. 70, 858–864.
- van Genuchten M.T., Wierenga P.J. (1976) Mass transfer studies in sorbing porous media. I. Analytical solutions, Soil Sci. Soc. Am. J. 40, 473–480.
- van Genuchten M.T., Wagenet R.J. (1989) Two-site/two-region for pesticide transport and degradation: theoretical development and analytical solutions, Soil Sci. Soc. Am. J. 53, 1303–1310.
- Vasileiadis V.P., Froud-Williams R.J., Eleftherohorinos I.G. (2007) Vertical distribution, size and composition of the weed seedbank under various tillage and herbicide treatments in a sequence of industrial crops, Weed Res. 47, 222–230.
- Wagner S.C., Zablotowicz R.M., Gaston L.A., Locke M.A., Kinsella J. (1996) Bentazon degradation in soil: Influence of tillage and history of bentazon application, J. Agric. Food Chem. 44, 1593–1598.
- Walker A. (1971) Effects of soil moisture content on the availability of soil-applied herbicides to plants, Pestic. Sci. 2, 56–59.
- Watts D.W., Hall J.K. (1996) Tillage and application effects on herbicide leaching and runoff, Soil Till. Res. 39, 241–257.
- Wauchope R.D. (1978) The pesticide content of surface water draining from agrigultural fields - a review, J. Environ. Qual. 7, 459–472.
- Weber J.B., Hardy D.H., Leidy R.B. (2002) Laboratory, greenhouse, and field lysimeter studies of <sup>14</sup>C-atrazine volatilization, in: Phelps W. et al. (Eds.), Pesticide environmental fate.

ACS Symp. Ser. 813. Am. Chem. Soc., Washington, DC, pp. 125–142.

- Weber J.B., Taylor K.A., Wilkerson G.G. (2006) Soil cover and tillage influenced metolachlor mobility and dissipation in field lysimeters, Agron. J. 98, 19–25.
- Webster E.P., Shaw D.R. (1996) Off-site runoff losses of metolachlor and metribuzin applied to differing soybean (Glycine max) production systems, Weed Technol. 10, 556–564.
- Weed D.A.J., Kanwar R.S., Stoltenberg D.E., Pfeiffer R.L. (1995) Dissipation and distribution of herbicides in the soil profile, J. Environ. Qual. 24, 68–79.
- Weed D.A.J., Kanwar R.S., Cambardella C., Moorman T.B. (1998) Alachlor dissipation in shallow cropland soil, J. Environ. Qual. 27, 767–776.
- Whang J.M., Schomburg C.J., Glotfelty D.E., Taylor A.W. (1993) Volatilization of fonofos, chlorpyrifos, and atrazine from conventional and no-till surface soils in the field, J. Environ. Qual. 22, 173–180.
- Wienhold B.J., Gish T.J. (1994) Effect of formulation and tillage practice on volatilization of atrazine and alachlor, J. Environ. Qual. 23, 292–298.

- Worsham A.D. (1991) Role of cover crops in weed management and water quality, in: Hargrove W.L. (Ed.), Cover Crops for Clean Water, Jackson, TN, USA, pp. 141–145.
- Xing B. (2001) Sorption of naphthalene and phenanthrene by soil humic acids, Environ. Pollut. 111, 303–309.
- Xing B., Pignatello J.J. (1997) Dual-mode sorption of lowpolarity compounds in glassy poly(vinylchloride) and soil organic matter, Environ. Sci. Technol. 31, 792–799.
- Xue S.L., Chen S., Selim H.M. (1997) Modeling alachlor transport in saturated soils from no-till and conventional tillage systems, J. Environ. Qual. 26, 1300–1307.
- Yuan G.S., Xing B.S. (1999) Site-energy distribution analysis of organic chemical sorption by soil organic matter, Soil Sci. 164, 503–509.
- Zablotowicz R.M., Locke M.A., Smeda R.J. (1998) Degradation of 2,4-D and fluometuron in cover crop residues, Chemosphere 37, 87–101.
- Zablotowicz R.M., Locke M.A., Gaston L.A., Bryson C.T. (2000) Interactions of tillage and soil depth on fluometuron degradation in a Dundee silt loam soil, Soil Till. Res. 57, 61–68.

# **Sustainable Cow-Calf Operations and Water Quality**

Gilbert C. Sigua

### Contents

1	Intro	oduction	834		
	1.1 Animal Numbers, Livestock Operations				
		and Animal Manure Production			
		in the United States	835		
	1.2	Overview: Cow-Calf Operation			
		and Management in Florida	836		
	1.3	Eutrophication Associated with Animal-Based			
		Agriculture: Overview	837		
2	Fora	ge-Based Cow-Calf Operation: Effect			
	on th	e Environment and Water Quality	839		
	2.1	Impact of Grazing Cattle, Cattle Movement,			
		and Grazing Behavior on Water Quality			
		and the Soil Nutrient Dynamics Around			
		and Beneath Cattle Congregation Sites	839		
	2.2	Effects of Forage Type and Harvest Method			
		on Above Ground Net Production, N and			
		P Uptake, and Soil C, P, and N Dynamics	040		
		in Subtropical Pastures	840		
3	Case	Studies: Effects of Grazing on Surface			
	Wate	er/Shallow Groundwater Quality and Soil			
	Qual	ity (Florida Experiences)	841		
	3.1	Case Study #1: Surface Water Quality	841		
	3.2	Case Study #2: Shallow Groundwater Quality	844		
	3.3	Changes in Soil P, K, Ca, Mg and pH	o <b>1</b> =		
		Associated with Cow-Calf Operation	847		
4	Man	agement Options: Managing Properly			
	Cow	-Calf Operations to Improve Water Quality			
	and l	Pasture Sustainability	847		
	4.1	Improving Pasture Sustainability	847		
5	Sum	mary and Conclusions	853		
Re	ferenc	res	853		

Abstract As animal agriculture has evolved to larger production operations in subtropical regions of United States, the problems associated with manure handling, storage and disposal have grown significantly. Understanding the interaction effects of sustainable cow farming with water-table management, nutrient dynamics and water quality in pastures could be the key to reducing nutrients in runoff. Soils do not contribute equally to nutrient export from watersheds or have the same potential to transport nutrient to runoff nor would soil test levels accurately predict total dissolved nutrients. Better understanding of soil nutrient dynamics and crop nutrient changes resulting from different management systems should allow us to predict potential impact on adjacent surface waters. In many states, these issues are critical and of increasing importance among environmentalists, ranchers, and public officials particularly in the case of N and P. One of the first steps in assessing N or P level on any farm is to consider total N or total P inputs and outputs. In Florida, reduction of P transport to receiving water bodies is the primary focus of several studies because P has been found to be the limiting nutrient for eutrophication in many aquatic systems. Long-term monitoring of the changes in soil nutrients, especially soil P would enable us to predict soil chemical or physical deterioration under continuous forage-livestock cultivation and to adopt measures to correct them before they actually happen. Despite substantial measurements using both laboratory and field techniques, little is known about the spatial and temporal variability of nutrient dynamics across the entire landscape, especially in agricultural landscapes with cow-calf operations.

G.C. Sigua (🖂)

Research Soil Scientist, United States Department of Agriculture-Agricultural Research Service Subtropical Agricultural Research Station, Brooksville, FL 34601, USA e-mail: gilbert.sigua@ars.usda.gov

**Keywords** Bahiagrass • Cow-calf • Groundwater • Surface water • Groundwater • Sustainability • Nutrient cycling • Plant uptake • Water quality • Subtropics • BMPs • Eutrophication • Trophic state index (TSI)

### **1** Introduction

Beef cattle (Bos taurus) pastures in subtropical regions of United States and other parts of the world are typically dominated by subtropical and tropical grasses such as bahiagrass (Paspalum notatum, Flügge) or bermudagrass (Cynodon dactylon, L). The establishment and maintenance of persistent grasslegume pastures is a key option to increase productivity and profitability of beef cattle production systems. The development of effective grass-legume pastures for beef cattle production is a good option to improve nutritional value of the pastures, minimize N or P fertilizers input, and better manage nutrient cycling to enhance water quality. The greatest environmental concern with many grazing areas in Florida is level of soil P due to P accumulation in soil, and the subsequent loss of sediment-bound and soluble P in runoff. Our ability to estimate the levels and changes of soil P and other crop nutrients in subtropical beef cattle pastures has the potential to improve our understanding of P dynamics and nutrient cycling at the landscape level.

Throughout the southeastern United States, grazinglands have considerable variability in soils, climate, and growing season, which not only affect the types of forage that can be grown, but also the overall environmental and biodiversity management. Grazing animals affect the movement and utilization of nutrients through the soil and plant system, and thus the fertility of pasture soils (Haynes and Williams, 1993). Different pasture species may affect nutrient use and turnover due to seasonal timing of growth, root type, and forage types (Haynes, 1981; Wedin and Tilman, 1990; Stout et al., 1997). Nutrient availability at watershed scale may control pasture growth and thus the number of domestic animals that can be supported. Increased nutrient availability, through fertilizer applications, and the subsequent increases in pasture production offer the potential for increased animal production (Haynes and Williams, 1993). Society relies on adequate freshwater resources to support populations of people, agriculture, industry, wildlife habitat, aquatic ecosystems, and a healthy environment. Consequently, the interaction of pasture management and hydrology is important issue to environmentalists, ranchers, and public officials because it may affect nutrient dynamics and water quality.

Forage-beef cattle research programs must adopt an integrated approach that will lead to the development of appropriate sustainable pasture technologies that optimize beef cattle ranching profitability. Thus, both actual and perceived environmental problems associated with beef cattle production systems need to be addressed when new management systems are being developed. A key issue to be evaluated is how different livestock management practices impact the environment, including water quality, flora and fauna bio-diversity, and soil and landscape integrity. Another equally important issue concerns the balance of fertility management for forage-livestock agro-ecosystem that may result in increased nutrient use efficiency and, therefore, less likelihood of nutrient loss to the environment due to leaching and/or runoff. Additionally, there is a heightened likelihood of P losses from over fertilized pastures through surface water runoff or percolation past the root zone (Gburek and Sharpley, 1998; Stout et al., 2000).

Reduction of P transport to receiving water bodies has been the primary focus of several studies because P has been found to be the limiting nutrient for eutrophication in many aquatic systems (Botcher et al., 1999; Sigua et al., 2000; Sigua and Tweedale, 2003). Elsewhere, studies of both large (Asmussen et al., 1975) and small watersheds (Romkens et al., 1973; Hubbard and Sheridan, 1983) have been performed to answer questions regarding the net effect of agricultural practices on water quality with time or relative to weather, fertility, or cropping practices.

Recent assessments of water quality status have identified eutrophication as one of the major causes of water quality "impairment" not only in the United States, but also around the world. In most cases, eutrophication has accelerated by increased inputs of P and/or N due to intensification of crop and animal production systems since the early 1990's. The current high demand for quality protein and fiber production because of increasing world population has resulted in an intensification of agricultural production systems. As animal-based agriculture has evolved



# Animal Units for Beef and Dairy Cattle, 1997

Fig. 1 Spatial distribution of animal units for beef and dairy cattle in the United States (Source: Kellog et al., 2000)

to larger production operations in subtropical region of United States, the problems associated with manure handling, storage and disposal have grown significantly. This review article examines the following two key questions: (1) are forage-based animal production systems as suggested by regulators the major sources of non-point source nutrients pollution that are contributing to the degradation of water quality in lakes, reservoirs, rivers, and ground water aquifers? and (2) is properly managed cow-calf operations in subtropical agro-ecosystem would not likely be the major contributors to excess loads of N or P in surface water and/or shallow groundwater?

# 1.1 Animal Numbers, Livestock Operations and Animal Manure Production in the United States

Data from the census of agriculture were used by Kellog et al. (2000) to make estimates of livestock populations in the United States. A census of agricul-

tural producers is being conducted every five years by the United States Department of Agriculture National Agricultural Statistics Service. The basic building block of the estimation process is an animal unit. For the purposes of this paper, an animal unit represents 1000 pounds or 450 kilograms of live animal weight. An example of animal units for beef and dairy cattle spatial distribution in the United States is shown in Figure 1. For more detailed information related to spatial and temporal trends of animal units in the United States, the published report by Kellog et al. (2000) is highly suggested. They also reported the estimated amount nutrients (i.e., N and P excreted) based on manure production associated for livestock animals units.

The amount of manure N and P excreted in the United States are shown in Figures 2 and 3. The annual amount of manure N and P production from beef cattle operations remained constant from 1982 to 1997 while manure N and manure P production for dairy cattle had shown slightly decreasing trend from 1982 to 1997. On the contrary, the manure N and manure P production from poultry operations showed a remarkable increasing trend from 1982 to 1997 (Figs. 2 and 3).

Fig. 2 Temporal distribution of manure nitrogen excreted in the United States (*Source: Kellog et al., 2000*)

**Fig. 3** Temporal distribution of manure phosphorus excreted in the United States (*Source: Kellog et al.*, 2000)



# 1.2 Overview: Cow-Calf Operation and Management in Florida

Eleven million ha of grazingland in the subtropical (23.5–30° N Lat) United States supports about 30% of the U.S. beef cow herd. Florida's beef production ranks 10th among beef producing states in the United States and 4th nationally among states in number of herds with more than 500 brood cows. Florida's beef cattle had sales of more than \$443 million in 2004. The majority of Florida's cow herd is located in the Kissimmee/Lake Okeechobee watershed, a place where the public is becoming more concerned about high levels of P entering the lake and subsequently flowing out into waterways and the Everglades.

Florida is a large state with a considerable variability in soils and climate. In north Florida, there are some clay-loam soils with good moisture-holding capacity that are quite productive (Chambliss et al., 2001). Coming down the peninsula, soils are dominated by sandy ridges and flatwoods. In general, the flatwood soils with their higher moisture-holding capacity are more productive than the upland deep, droughty sands. The warm growing season is longer in south Florida than in north Florida while winter temperatures are usually lower in north Florida than in south Florida. These differences in climate, soils and length of growing season affect the types of forage that can be grown. Nevertheless, Florida's relatively mild climate, together with more than 127 cm of annual rainfall, affords a better opportunity for nearly 12 months of grazing than in any other state except Hawaii (Chambliss et al., 2001).

1987

1992

YEARS

1997

### 1.2.1 Pasture Management: Grazing and Fertilization (Typical in Florida Ranches)

1982

Bahiagrass is a common pasture used for beef cattle across Florida. Fertility and management practices have been based on University of Florida's

	Average days grazed per pasture	Animal unit per month <sup>1</sup>	Total feces excreted <sup>2</sup>	Total feces nitrogen excreted <sup>3</sup>	Total urine nitrogen excreted <sup>4</sup>	Total nitrogen after losses <sup>5</sup>		
Months	$kg ha^{-1} month^{-1}$							
January	13.8	1.0	913	5.0	2.8	2.3		
February	9.4	0.8	669	3.7	2.0	1.7		
March	13.5	0.9	753	4.1	2.5	1.9		
April	12.0	0.7	619	3.4	1.9	1.6		
May	12.7	0.7	654	3.6	1.9	1.7		
June	12.0	0.9	748	4.1	2.5	1.9		
July	6.9	0.7	628	3.4	1.9	1.6		
August	9.1	0.8	734	4.0	2.5	1.9		
September	8.2	1.1	947	5.2	3.0	2.5		
October	6.7	1.1	976	5.3	3.1	2.5		
November	6.6	0.8	705	3.9	2.2	1.8		
December	9.4	1.2	1037	5.7	3.4	2.7		
Totals	-	10.8	9347	51.2	27.8	24.1		

Table 1 Monthly grazing activity and estimates of nitrogen contributions from cattle feces and urine (Source: Sigua et al., 2009)

<sup>1</sup>Animal units per month (450 kg cow/calf unit).

<sup>2</sup>Total feces excreted (kg as excreted) = [(number of AUM × total annual animal feces excretion/12) × total feces excretion (as excreted) per animal per year] = 10.4 metric tons (Kellog et al., 2000).

<sup>3</sup>Total nitrogen excreted = total feces excreted × percent total nitrogen in feces (0.55%; Kellog et al., 2000). Total urine nitrogen excreted = based on urine N (g day per animal with live weight of 381 kg). Values on this table were adjusted for 450 kg cow-calf unit (Yan et al., 2007).

<sup>4</sup>Total nitrogen after losses = based on volatilization correction of 70% (during and after animal excretion).

recommendations as described by Chambliss (1999). Pastures are being grazed during spring of the year. After the start of summer rainy season, pastures that are to be hayed are being dropped out of the grazing cycle (usually starting in July). Pasture fields with bahiagrass are normally fertilized in the spring with 90 kg N ha<sup>-1</sup> and 45 kg K<sub>2</sub>O ha<sup>-1</sup>. Grazing cattle at the United States Department of Agriculture Subtropical Agricultural Research Station in Brookville, FL and other Florida ranches are being rotated among pastures on a 3-day grazing interval with 24 days of rest between pastures. The average number of grazing cattle was about 2.9 animal units per hectare and grazing days of about 5.5 on monthly basis as shown in Table 1. In addition, the number of days grazed each month, average number of animals per hectare and estimated total feces excreted along with the estimated total N in feces and from urine are also shown in Table 1.

# 1.3 Eutrophication Associated with Animal-Based Agriculture: Overview

Eutrophication is frequently a result of nutrient pollution, such as the release of sewage effluent, urban stormwater run-off, and run-off carrying excess fertilizers into natural waters. Nutrients (e.g., N or P) and other pollutants may enter from a number of sources (Fig. 4). However, it may also occur naturally in situations where nutrients accumulate (e.g. depositional environments) or where they flow into systems on an ephemeral basis. Eutrophication generally promotes excessive plant algal growth and decay, favors certain weedy species over others, and is likely to cause severe reductions in water quality (Schindler, 1974).

In aquatic environments, enhanced growth of choking aquatic vegetation or phytoplankton (that is, an algal bloom) disrupts normal functioning of the ecosystem, causing a variety of problems such as a lack of oxygen in the water, needed for fish and shellfish to survive (Fig. 5). The water then becomes cloudy, colored a shade of green, yellow, brown, or red. Human society is impacted as well; eutrophication decreases the resource value of rivers, lakes, and estuaries such that recreation, fishing, hunting, and aesthetic enjoyment are hindered. Health-related problems can occur where eutrophic conditions interfere with drinking water treatment. Many drinking water supplies throughout the world may experience periodic massive surface blooms of cyanobacteria (Kotak et al., 1993). These blooms contribute to a wide range of water-related



Fig. 4 Nutrient losses from farm gate on watershed scale causing eutrophication



Fig. 5 Effect of accelerated eutrophication from farm gate to harmful algal blooms and fish kills due to hypoxia

problems including summer fish kills (Fig. 5), and unpalatability of drinking water (Palmstrom et al., 1988).

Increased loss of nutrients in agricultural runoff has potentially serious ecological and public health implications (Hooda et al., 2000). Nitrogen and P are particularly important as both are implicated in aquatic eutrophication (Levine and Schindler, 1989). Eutrophication and the associated ecological effects result in a general decline in overall water quality, restricting its use for general and drinking purposes (USEPA, 1988; Sharpley and Withers, 1994).

# 2 Forage-Based Cow-Calf Operation: Effect on the Environment and Water Quality

Beef cattle operations have been suggested as one of the major sources of non-point source P and N pollution that are contributing to the degradation of water quality in lakes, reservoirs, rivers, and ground water aquifers in Florida (Allen et al., 1976, 1982; Bogges et al., 1995; Edwards et al., 2000). Cattle manure contains appreciable amounts of N and P (0.6% and 0.2%, respectively), and portions of these components can be transported into receiving waters during severe rainstorms (Khaleel et al., 1980). Work in other regions of the country has shown that when grazing animals become concentrated near water bodies, or when they have unrestricted long-term access to streams for watering, sediment and nutrient loading can be high (Thurow, 1991; Brooks et al., 1997). Additionally, there is a heightened likelihood of N and P losses from overfertilized pastures through surface water runoff or percolation past the root zone (Schmidt and Sturgul, 1989; Gburek and Sharpley, 1998; Stout et al., 1998, 2000).

Reduction of P transport to receiving water bodies has been the primary focus of several studies because P has been found to be the limiting nutrient for eutrophication in many Florida aquatic systems (Botcher et al., 1999; Sigua et al., 2000; Sigua and Steward, 2000; Sigua and Tweedale, 2003). Recently, Sigua et al. (2004, 2006) found that the levels of soil P varied widely with different pasture management. Water quality in lakes associated with cattle production was "good", equivalent to 30-46 trophic state index (TSI) based upon the Florida Water Quality Standard. These findings indicate that properly managed livestock operations may not be major contributors to excess loads of nutrients (especially P) in surface water (Sigua et al., 2006). In another study in south Florida, Arthington et al. (2003) reported that the presence of beef cattle at three stocking rates (1.5, 2.6, and 3.5 ha per cow) had no impact on nutrient loads (P and N) in surface runoff water compared with pastures containing no cattle. However, these studies should not be considered as definitive for the region because of the wide range in management options (fertilization, stocking rate, grazing system, forage type, etc.) that are used in beef cattle production systems. Whether or not nutrient losses from grazed pastures are significantly greater

than background losses and how these losses are affected by soil, forage management, or stocking density are not well defined (Gary et al., 1983; Edwards et al., 2000; Sigua et al., 2004). Concern for losses of soil P by overland flow were noted when soil P exceeded 150 mg kg<sup>-1</sup> in the upper 20-cm of soil (Johnson and Eckert, 1995; Sharpley et al., 1996). Sharpley (1997) noted that all soils do not contribute equally to P export from watersheds or have the same potential to transport P to runoff. In their studies, Coale and Olear (1996) observed that soil test P levels did not accurately predict total dissolved P. Better understanding of soil P dynamics and other crop nutrient changes resulting from different management systems should allow us to better predict potential impact on adjacent surface waters. These issues are critical and of increasing importance among environmentalists, ranchers, and public officials in the state (Sigua et al., 2006).

# 2.1 Impact of Grazing Cattle, Cattle Movement, and Grazing Behavior on Water Quality and the Soil Nutrient Dynamics Around and Beneath Cattle Congregation Sites

Understanding cattle movement in pasture situations is critical to assess their impact on agro-ecosystems. Movement of free-ranging cattle varies due to spatial arrangement of forage resources within pastures (Senft et al., 1985) and the proximity of water (Holechek, 1988; Ganskopp, 2001), minerals (Martin and Ward, 1973), and shade to grazing sites. The breed of the animal also affects livestock distribution pattern (Herbel et al., 1967). Hammond and Olson (1994) and Bowers et al. (1995) reported that temperate British breeds (Angus and Hereford) of Bos taurus cows grazed less during the day than tropically adapted Senepol cows, but compensated for reduced grazing activity during the hotter parts of the day by increasing time spent grazing at night. Grazing animals congregate close to the shade and watering areas during the warmer periods of the day (Mathews et al., 1994, 1999). White et al. (2001) claimed that there was a correlation between time spent in a particular area and the number of excretions and this behavior could lead to an increase in the concentration of soil nutrients close to shade and water. Sigua (2004) demonstrated that concentrations of total inorganic N, total P and the degree of soil compaction varied significantly among different animal congregation sites. The highest concentrations of total inorganic N and total P were found at the shade and mineral feeder sites, respectively. The most compacted soil was at the mineral feeders' site. Although the levels of total inorganic N and total P were high near the center of the congregation sites, their levels did not increase with soil depth and their concentrations decreased almost linearly away from the center of the congregation sites. Soil compaction tended to decrease away from the center of the mineral feeder sites, but not at the water trough or shade sites. This study suggest that congregation sites in beef cattle operations in Florida are not as nutrient rich as suspected, and may not contribute more nutrients to surface and groundwater supply under Florida conditions because P levels at the center of sites were below 150 mg kg<sup>-1</sup> (Sigua, 2004). This concentration of soil P should not be considered an absolute maximum number for soil P to become harmful to water quality and the environment, but rather a good indicator of P accumulation in the soil. Furthermore, since there was no evidence of a vertical build up or horizontal movement of inorganic N or total P in the landscape, Sigua et al. (2005) surmised that cattle congregation sites may not be considered a substantial source of nutrients at the watershed level.

Grazing animals impact the movement and cycling of nutrients through the soil and plant system, and thus on the fertility of pasture soils (Haynes, 1981; Haynes and Williams, 1993). Grazing can accelerate and alter the timing of nutrient transfers, and increase amounts of nutrients cycled from plant to soil (Klemmendson and Tiedemann, 1995). Long periods of time, position of shade and water resources for grazing cattle can influence the spatial distribution of soil biochemical properties including soil organic C and N, particulate organic C and N, microbial biomass, and net N mineralization (Ruess and McNaughton, 1987; Kieft, 1994; Kieft et al., 1998; Franzluebbers, 2000). Long-term intensive grazing may decrease the input of organic matter into soils in the immediate vicinity of individual plants and eventually reduce nutrient concentrations beneath plants by limiting availability of photosynthesis and/or meristematic tissues necessary for growth (Milchunas and Lauenroth, 1993; Briske and Richards, 1995). Thrash (1997) reported that concentration of large herbivores around the troughs negatively

impacted the infiltration rate of the soils, with implications for the rate of soil loss and the soil moisture regime. Elsewhere, grazing, trampling, and dung deposition by large herbivores often result in a zone of decreasing impact on many vegetation and soil parameters including herbaceous vegetation basal cover, soil bulk density, and penetrability away from water points (Andrew and Lange, 1986; Thrash et al., 1991). The effect of trampling appears to be less severe on vegetated grasslands than on poor or bare soil (Warren et al., 1986). Studies on grazing and soil compaction generally find that exposure to livestock grazing compacts soil and that soil compaction increases with grazing intensity. This pattern is reflected in reviews of the scientific literature on the subject (Fleischner, 1994; Lauenroth et al., 1994; Kauffman and Krueger, 1984).

# 2.2 Effects of Forage Type and Harvest Method on Above Ground Net Production, N and P Uptake, and Soil C, P, and N Dynamics in Subtropical Pastures

The productivity of any ecosystem depends on the amounts of nutrients stored in various compartments, such as vegetation, litter, soil, and animal biomass, and on the rates of nutrient cycling and transfer among those compartments. The cycling of nutrients in a given ecosystem is affected by a combination of biological and physical processes (Holt and Coventry, 1990). The relative importance of these processes varies considerably between ecosystems as a result of differences between climate, soils, vegetation, and management practices. Nutrient dynamics in various agro-ecosystems are continually evolving in response to changing management practices. Soil dynamics may continue to change in response to external abiotic perturbations such as global changes in temperature, precipitation, and CO<sub>2</sub> concentration (Janzen et al., 1997, 1998).

Different pasture species affect nutrient use and turnover due to seasonal timing of growth, root type, and forage types (Wedin and Tilman, 1990; Stout et al., 1997). Other researchers (Anderson and Coleman, 1985; Dormaar, 1992) also have attributed the amount of grassland soil organic matter to the amount of root biomass. Vegetation grazed by livestock is rapidly



Main Station: Forage-Based Pasture

Fig. 6 Location of the study sites and aerial view of the Subtropical Agricultural Research Station, Brooksville, Hernando County, Florida, USA

decomposed during digestion and many nutrients are returned to the soil in readily available forms in feces and urine (Lauenroth et al., 1994).

# 3 Case Studies: Effects of Grazing on Surface Water/Shallow Groundwater Quality and Soil Quality (Florida Experiences)

### 3.1 Case Study #1: Surface Water Quality

### 3.1.1 Experimental Methods

The lakes that we studied were adjacent to or within about 14-km away from the USDA-ARS, Subtropical Agricultural Research Station (STARS), Brooksville, FL (Fig. 6). These lakes are associated with foragebased beef cattle operations. The lakes were (1) Lake Lindsey (28°37.76'N, 82°21.98'W), adjacent to STARS; (2) Spring Lake (28°29.58'N; 82°17.67'W), about 10 km away from STARS; and (3) Bystre Lake (28°32.62'N; 82°19.57'W), about 14 km away from STARS.

Monthly water quality monitoring of lakes associated with beef cattle pastures was begun in 1993 and continued until 2002 by the field staff of the Southwest Florida Water Management District (SWFWMD). Monthly water samples were taken directly from the lakes using a water (Van Dorn) grab sampler. Water quality parameters monitored were Ca, Cl,  $NO_2 +$  $NO_3$ -N,  $NH_4$ -N, total N, total P, K Mg, Na, Fe, and pH. All sampling, sample preservation and transport, and chain of custody procedures were performed in accordance with an EPA-approved quality assurance plan with existing quality assurance requirements (USEPA,

1979; APHA, 1992). The SWFWMD Analytical Laboratory, using EPA-approved analytical methods, performed the chemical analyses of water samples from the lakes (USEPA, 1979).

Trophic state index development and calculation. Lake trophic state index (TSI) is understood to be the biological response of a lake to forcing factors such as nutrient additions. Nutrients promote growth of microscopic plant cells (phytoplankton) that are fed upon by microscopic animals (zooplankton). The TSI of (Carlson, 1983) uses algal biomass as the basis for trophic state classification (e.g. oligotrophic, mesotrophic, eutrophic, and hypereutrophic).

The Florida trophic state index of (Brezonik, 1984), which was derived using data from 313 Florida Lakes, was modified in the study. The first step involved in assigning a TSI value was to assess the current nutrient status based on TN/TP ratio of the lake. The TN to TP ratios were classified into three categories namely: N limited (TN/TP < 10:1); P limited (TN/TP > 30:1); and balanced (10:1  $\leq$  TN/TP  $\leq$  30:1). The TSI of each lake that we studied was calculated by entering key water quality parameters: total P (TP, ug  $L^{-1}$ ); total N (TN, mg  $L^{-1}$ ); chlorophyll a (CHL, mg  $m^{-3}$ ) for the measurements of planktonic algae density, and Secchi depth (SD, m) for measuring water transparency into an empirical formula (Eqs. (1) to (3)). Equation (1)(P-limited) was used to calculate the TSI values for Lake Lindsey and Spring Lake while equation (2) (Nlimited) was used to calculate the TSI value for Bystere Lake. A novel paper on trophic state index for lakes written by (Carlson, 1977) was an excellent reference to explain the mathematical derivations of the different equations listed below.

### **I. PHOSPHORUS - LIMITED LAKES**

(TN/TP > 30:1)

(1)TSI(AVG) = 1/3[TSI(CHL) + TSI(SD) + TSI(TP)]where: TSI (CHL) =  $16.8 + 14.4 * \ln$  (CHL), (mg  $m^{-3}$ )  $TSI(SD) = 60.0 - 30.0 * \ln(SD), (m)$ 

 $TSI (TP) = -23.8 + 23.6 * \ln (TP), (ug L^{-1})$ 

# **II. NITROGEN - LIMITED LAKES**

(TN/TP < 10:1)(2)TSI (AVG) = 1/3 [TSI (CHL) + TSI (SD) + TSI(TN)] where: TSI (CHL) =  $16.8 + 14.4 * \ln$  (CHL), (mg  $m^{-3}$ )

- $TSI(SD) = 60.0 30.0 * \ln(SD), (m)$  $TSI (TN) = 59.6 + 21.5 * \ln (TN), (ug L^{-1})$
- **III. NUTRIENT BALANCED LAKES**  $(10:1 \leq TN/TP \leq 30:1)$ (3) TSI (AVG) = 1/3 [TSI (CHL) + TSI (SD) + (0.5 TSI)(TP) + TSI(TN))] where: TSI (CHL) =  $16.8 + 14.4 * \ln$  (CHL), (mg  $m^{-3}$ )  $TSI(TN) = 56 + 19.8 * \ln(TN), (ug L^{-1})$  $TSI (TP) = -18.4 + 18.6 * \ln (TP), (ug L^{-1})$  $TSI(SD) = 60.0 - 30.0 * \ln(SD), (m)$

### 3.1.2 Highlights of Research Results

Status of water quality in lakes. Assessment of water quality data from 1993 to 2002 confirms that water-quality variations (temporal and spatial) existed in lakes associated with beef cattle pasture systems in Central Florida. The lakes were found to differ from each other in Ca, NO<sub>2</sub>+NO<sub>3</sub>-N, TN, TP, K, Mg, Na, and Fe. Significant temporal variations were observed for NH<sub>4</sub>-N, TP, and Fe while significant interaction effects (lakes  $\times$  year) were only noted for NH<sub>4</sub>-N, TP, Mg, Na, and Fe.

The levels of NO<sub>2</sub>+NO<sub>3</sub>-N and NH<sub>4</sub>-N in lakes did not show any significant differences from 1993 to 2002 while TN of Bystere Lake declined from 1.12 to  $0.76 \text{ mg L}^{-1}$  between 1993 and 2002. Lake Lindsey's TN in 1993 was about 0.82 mg  $L^{-1}$  and 0.80 mg  $L^{-1}$ in 2002 while Spring Lake's TN in 1993 and 2002 were 0.65 and 0.78 mg  $L^{-1}$ , respectively. The levels of TP in Bystere Lake between 1993 and 2002 increased from 0.08 to 0.34 mg  $L^{-1}$  while levels of TP in Lake Lindsey did not change from 1993 to 2002. A decline of TP was noted in Spring Lake from 1993 (0.19 mg  $L^{-1}$ ) to 2002 (0.01 mg  $L^{-1}$ ). With the continuous conversion of cropland and pastureland to residential use (although at slow pace), contribution of nutrients from anthropogenic sources is becoming a big concern environmentally over time for lakes associated with foragebased pasture systems.

Total nitrogen (TN)/total phosphorus (TP) ratio. Nitrogen and P are the primary crop nutrients that can impact the environment. When applied in excess of crop needs, nutrients can run off into surface waters resulting in excessive aquatic plant growth and toxicity



**Fig. 7** Total nitrogen (TN) to total phosphorus (TP) ratio of lakes associated with forage-based beef cattle pasture system (*Source: Sigua et al., 2006c*)

to certain fish species. The TN/TP ratio may be a useful method to establish the N and P reduction targets in the environment (Sigua et al., 2000). The ratio of TN to TP is one of the important components in calculating the TSI of lakes.

Several studies have shown that a TN/TP ratio  $\leq 10:1$  appears to favor algal blooms, especially blue-green algae, which are capable of fixing atmospheric N (Schindler, 1974; Chiandini and Vighi, 1974; Sakamoto, 1966; Sigua et al., 2000). Figure 7 shows the TN/TP ratio of the lakes that were associated with beef cattle operations. Lake Lindsey and Spring Lake can be classified as P-limited lakes with TN/TP ratios of 41:1 and 51:1, respectively. Bystere Lake with a TN/TP ratio of 9:1 was classified as an Nlimited lake and may have higher probability for algal bloom compared with Lake Lindsey and Spring Lake because of its higher P levels. From 1993 to 2002, the TP in Bystere Lake increased from 0.08 mg L<sup>-1</sup> to 0.34 mg L<sup>-1</sup>.

**Trophic state index of lakes (TSI).** The Florida TSI was devised to integrate different but related measures of lake productivity or potential productivity, into a single number that ranges from 0 to 100. The measures included in the calculation of TSI are water transparency (Secchi depth), chlorophyll *a* (measurement of algae content), TN, and TP. The Florida TSI for Lake Lindsey, Spring Lake, and Bystre Lake were 35, 30, and 46, respectively (Fig. 8). Based on this, the TSI of these lakes can be classified as "good" according to Florida water quality standard (TSI of 0–59 = "good"; TSI of



**Fig. 8** Trophic state index for lakes with forage-based beef cattle pasture system. Trophic state index is sginficantly different ( $P \le 0.05$ ) when superscripts located at top bars are different (*Source: Sigua et al., 2006c*)

60 to 69 = "fair"; and TSI of 70 to 100 = "poor"). Although the TSI levels of the three lakes did not show any significant change from 1993 to 2002, TSI levels increased numerically for all lakes (Fig. 8). This is reflected in a change in the trophic status of Bystere Lake. Lake Lindsey with TSI of 31 and 38 in 1993 and 2002, respectively, remained within the mesotrophic classification, while Spring Lake with TSI of 25 and 26 in 1993 and 2002, respectively, remained in the oligotrophic category. Lake Lindsey (mesotrophic lake) would normally have moderate nutrient concentrations with moderate growth of algae and/or aquatic macrophytes and with clear water (visible depth of 2.4 to 3.9 m).

Oligotrophic lake such as Spring Lake would normally have less abundance of aquatic macrophytes and algae, or both because nutrients are typically in short supply. Oligotrophic lakes tend to have water clarity greater than 3.9 m due to low amounts of freefloating algae in the water column.

Bystere Lake, which was at the upper end of the mesotrophic range in 1993 (TSI of 49), shifted into the slightly eutrophic state in 2002 with a TSI value slightly above 50. Eutrophic lakes normally have green, cloudy water, indicating lots of algal growth in the water. Water clarity of most eutrophic lakes generally ranges from 0.9 to 2.4 m. Generally, water quality in Lake Lindsey and Spring Lake was consistently good (1993–2002) while water quality of Bystere Lake ranged from good in 1993 to fair in 2002 (Fig. 8).

# 3.2 Case Study #2: Shallow Groundwater Quality

### 3.2.1 Experimental Methods

Instrumentation and water sample collection. Two adjacent 8-ha pasture fields were instrumented with a pair of shallow wells placed at different landscape positions. The different landscape positions are top slope (TS; 10-20% slope, 2 ha; middle slope (MS; 5–10% slope, 2 ha and bottom slope (BS; 0–5% slope, 2 ha). The wells were constructed of 5 cm schedule 40 PVC pipe and had 15 cm of slotted well screening at the bottom. During installation of wells, sand was placed around the slotted screen, and bentonite clay was used to backfill to the soil surface to prevent surface water or runoff from moving down the outside of the PVC pipe and contaminating groundwater samples. A centralized battery-operated peristaltic pump was used to collect water samples. Wells were completely evacuated during the sampling process to ensure that water for the next sampling would be fresh groundwater. Water samples were collected from the groundwater wells every two weeks. However, there were periods when ground water levels were below the bottom level of the wells and samples could not be obtained. In addition to ground water samples, surface

water samples were collected in the pasture bottoms or the seep area when present, by taking composite grab samples on the same schedule. The seep area, which is located at the lower end of BS is a remnant of a sinkhole formation and became a small scale lake with varying levels of surface water. The seep area of about 2 ha in size is where runoff and seepage from higher parts of pasture converge.

Water sample handling and analyses. Water samples were transported to the laboratory following collection and refrigerated at 4°C. Water samples were analyzed for NO<sub>3</sub>-N and NH<sub>4</sub>-N using a Flow Injector Analyzer according to standard methods (APHA, 1989).

### 3.2.2 Highlights of Research Results

Concentration of NH<sub>4</sub>-N, NO<sub>3</sub>-N and TIN in Surface and Ground Water

Concentrations of NH<sub>4</sub>-N, NO<sub>3</sub>-N, and TIN in shallow groundwater did not vary with landscape positions (Fig. 9). However, concentrations NH<sub>4</sub>-N, NO<sub>3</sub>-N, and TIN in the water samples collected from the seep area were significantly ( $P \le 0.05$ ) higher when compared to their average concentrations in water



Fig. 9 Average concentrations of  $NH_4$ -N,  $NO_3$ -N and TIN in shallow groundwater at different landscape positions. Line above the bars and across the line represents standard error of the mean. Means of  $NH_4$ -N (a & b),  $NO_3$ -N(x & y) and

TIN (m & n) in shallow groundwater are significantly different ( $P \leq 0.05$ ) when superscript located at top bar and line is different (*Source: Sigua et al., 2009a*)

Table 2   Summary statistics	Statistical parameters	NO <sub>3</sub> -N	NH <sub>4</sub> -N	Total inorganic N		
for the concentration of	2004					
groundwater under babiagrass	Number of samples (n) (N)	40	40	40		
pastures associated with	Mean (mg $L^{-1}$ )	0.7	0.1	0.8		
cow-calf operations	Median (mg $L^{-1}$ )	0.4	0.04	0.6		
-	Mode (mg $L^{-1}$ )	0.2	0.04	0.2		
	Maximum (mg $L^{-1}$ )	4.7	1.3	5.9		
	Minimum (mg $L^{-1}$ )	0.2	0.03	0.2		
	Std. error mean	0.2	0.03	0.2		
	Variance	0.9	0.04	0.9		
	Skewness	3.2	4.5	3.2		
		2005				
	Number of samples (n)	45	45	45		
	Mean (mg $L^{-1}$ )	0.8	0.3	1.03		
	Median (mg $L^{-1}$ )	0.3	0.1	0.7		
	Mode (mg $L^{-1}$ )	0.1	0.1	0.3		
	Maximum (mg $L^{-1}$ )	4.1	2.8	6.9		
	Minimum (mg $L^{-1}$ )	0.1	0.1	0.2		
	Std. error mean	0.1	0.1	0.1		
	Variance	0.8	0.2	1.0		
	Skewness	2.1	4.6	2.1		
		<u>2006</u>				
	Number of samples (n)	10	10	10		
	Mean (mg $L^{-1}$ )	0.7	0.2	0.8		
	Median (mg $L^{-1}$ )	0.4	0.1	0.6		
	Mode (mg $L^{-1}$ )	-	0.1	_		
	Maximum (mg $L^{-1}$ )	2.7	0.4	3.1		
	Minimum (mg $L^{-1}$ )	0.1	0.1	0.2		
	Std. error mean	0.2	0.03	0.2		
	Variance	0.6	0.01	0.6		
	Skewness	0.8	1.9	1.8		

Source: Sigua et al., 2009a.

samples collected from the different landscape positions (Fig. 9). Averaged across year, concentration of TIN ranged from 0.5 to 1.5 mg L<sup>-1</sup>. The highest TIN concentration occurred ( $P \le 0.05$ ) in the surface water while the concentrations from the shallow groundwater wells (BS-0.6 mg L<sup>-1</sup>, MS-0.9 mg L<sup>-1</sup>, and TS-0.6 mg L<sup>-1</sup>) were similar to each other and lower than the seepage area (Fig. 9).

Average concentrations of NO<sub>3</sub>-N (0.4 to 0.9 mg L<sup>-1</sup>) among the different sites were well below the maximum, of 10 mg L<sup>-1</sup>, set for drinking water (Fig. 9). On the average, the concentrations of NO<sub>3</sub>-N did not vary significantly with landscape positions, and as with TIN, the levels were significantly lower than surface water from seepage area (Fig. 9). The maximum NO<sub>3</sub>-N concentrations (averaged across landscape position) in shallow groundwater for 2004, 2005 and 2006 were also below the drinking

water standards for NO<sub>3</sub>-N. Other summary statistics for the levels of NO<sub>3</sub>-N, NH<sub>4</sub>-N and TIN in shallow groundwater are shown in Table 2.

Similar trends in landscape positions were found for average concentrations of NH<sub>4</sub>-N (Fig. 9). Again, the concentrations of NH<sub>4</sub>-N in shallow groundwater did not vary significantly among top slope, middle slope, and bottom slope wells. These levels of NH<sub>4</sub>-N were lower than that of the surface water (0.5 mg L<sup>-1</sup>). Annual average concentrations of NO<sub>3</sub>-N, NH<sub>4</sub>-N and TIN in shallow groundwater of pastures associated with beef cattle operations did not vary significantly with the different year of sampling. The highest concentration of TIN was in 2005 followed by 2006 and 2004 (Table 2).

Average concentrations of NO<sub>3</sub>-N in surface water and shallow groundwater (0.4 to 0.9 mg  $L^{-1}$ ) among the different sites did not exceed the drinking water standard for NO<sub>3</sub>-N (10 mg L<sup>-1</sup>). Nitrate levels in excess of 10 mg L<sup>-1</sup> in drinking water can cause health problems for human infants, infant chickens and pigs, and both infant and adult sheep, cattle and horses. The relatively constant concentration of NO<sub>3</sub>-N in the shallow groundwater over the three years (2004–2006) could be due to the combined effects of precipitation, N fertilization, and leaching from plant decomposition and animal feces and urine. Uptake of N by the actively growing bahiagrass reduced the amount of mineral N remaining vulnerable to leaching during the growing season (Decau et al., 2003).

# Concentration of Total Phosphorus and Degree of Phosphorus Saturation in Soils

Concentrations of total P in soils varied significantly  $(P \le 0.001)$  with landscape position and sampling depth, but there was no interaction effect of landscape position and sampling depth (Table 3). Soil samples from the seep area had the lowest concentration of total P when compared with other landscape positions. Soils from the middle slope  $(9.2 \pm 1.8 \text{ mg kg}^{-1})$  had the greatest concentration of total P followed by top slope  $(5.9 \pm 1.8 \text{ mg kg}^{-1})$  and bottom slope  $(5.7 \pm 1.8 \text{ mg kg}^{-1})$ 

**Table 3** Average concentration ( $\pm$  std. error of mean) and F-values of total phosphorus and degree of phosphorus saturation in soils at various landscape positions and soil depths 1.5 mg kg<sup>-1</sup>). Averaged across years, total P in the soil was about 9.1 mg kg<sup>-1</sup>.

Degree of P saturation in the soils varied significantly ( $P \leq 0.001$ ) with landscape position and sampling depth, but was not affected significantly by the interaction of landscape position and sampling depth (Table 3). The middle slope position (19.9 ± 4.9 %) had the highest degree of P saturation followed by top slope, bottom slope and seep area. Soils collected at sampling depth of 0–20 cm (20.9 ± 6.1 %) had significantly higher degree of P saturation than soils collected between 20 and 100 cm.

There was a significant ( $P \leq 0.05$ ) decrease in the average concentrations of total P with increasing sampling depth (Table 3). The upper two depths (0– 20 cm and 20–40 cm) had the highest concentrations while the lowest amount of total P was found in the lowest sampling depth of 60–100 cm (Table 3). These results suggest that there had been little movement of total P into the soil pedon since average degree of P saturation in the upper 20 cm was 21% while degree of P saturation at lower soil depth (60–100 cm) was about 3%.

Results indicate that current pasture management including cattle rotation in terms of grazing days and current fertilizer (inorganic + manures + urine)

Soil	Total	Degree of	
parameters	phosphorus	phosphorus saturation	
	- mg kg <sup>-1</sup>	%	
A. Landscape position			
1. Top slope	$5.91 \pm 1.77b^{*}$	$14.79 \pm 4.57a$	
2. Middle slope	$9.19 \pm 1.77a$	$19.92 \pm 4.97a$	
3. Bottom slope	$5.67 \pm 1.53b$	$7.86 \pm 2.14b$	
4. Seep area	$0.38 \pm 0.13c$	$1.25 \pm 0.73c$	
LSD(0.05)	2.62	6.35	
B. Soil depth (cm)			
1.0-20	$7.05 \pm 1.86a$	$20.93 \pm 6.11a$	
2.20-40	$9.05 \pm 2.18a$	$13.77 \pm 3.22b$	
3.40-60	$3.24 \pm 0.94b$	$5.64 \pm 1.67c$	
4.60–100	$1.81 \pm 0.62b$	$3.47 \pm 1.06c$	
LSD(0.05)	2.62	6.35	
	<b>F-values</b>	<b>F-values</b>	
Landscape position (LP)	17.37***	14.77**	
Soil depth (SD)	14.66**	14.24**	
$LP \times SD$	2.13 <sup>ns</sup>	2.28 <sup>ns</sup>	

\*Means in columns within each subheading followed by common letter(s) are not significantly different from each other at  $P \le 0.05$ . \*\*Significant at  $P \le 0.001$ ; ns – not significant. application rates for bahiagrass pastures offer little potential for negatively impacting the environment. Properly managed livestock operations contribute negligible loads of total P to shallow groundwater and surface water. Overall, there was no buildup of soil total P in bahiagrass-based pasture. Therefore, results of this study may help to renew the focus on improving fertilizer efficiency in subtropical beef cattle systems, and maintaining a balance of P removed to P added to ensure healthy forage growth and minimize P runoff.

## 3.3 Changes in Soil P, K, Ca, Mg and pH Associated with Cow-Calf Operation

During the last 15 years (1988–2002), concentrations for P, K, Ca, Mg, and soil pH have declined by about 7%, 38%, 46%, 61%, and 23% in pastures with bahiagrass and 27%, 55%, 76%, 56%, and 22% for pasture fields with bahiagrass + Rhizoma peanuts that were grazed in spring and hayed in early fall, respectively (Table 4). However, the levels of Ca:Mg ratio in fields with grazed and hayed bahiagrass and grazed and hayed Rhizoma peanuts had increased by about 42% and 29%, respectively. It is worth mentioning that depletion rates of P, K, Ca, and Mg were greater in pasture fields with bahiagrass suggesting that the former has greater nutritional demands.

Long-term monitoring on the changes in soil P and other crop nutrients in subtropical foraged-based beef

**Table 4** Changes (%) in Mehlich-1 extractable soil P and other nutrients under different pasture management in STARS, Brooksville, FL in 2002 relative to the 1988 levels of Mehlich-1 extractable soil P and other nutrient (*Source: Sigua et al., 2006b*)

	Pasture management			
	BG+GZHY	RP-G+GZHY		
Soil nutrients	(%)	(%)		
Р	-6.6	-26.9	_	
K	-38.2	-55.1		
Ca	-45.8	-75.6		
Mg	-61.2	-56.0		
Ca/Mg ratio	+42.1	+29.0		
рН	-22.1	-21.9		

BG – bahiagrass; RP-G – rhizoma peanuts-mixed grass; GZHY – grazing + haying.

cattle pastures enabled to predict soil chemical deterioration and/or soil accumulation of nutrients that could occur under continuous forage-livestock cultivation and to adopt measures to correct them before they actually happen. In addition, long-term monitoring also provided answers about the nutrient dynamics/cycling and turnover in legume-grass mixtures and the efficiency of fertilizer use. The knowledge that was gained on the relationship of temporal and spatial changes in soil nutrient levels in forage-based beef cattle pasture should indeed provide insights for improved grazing management, which could be both economically and environmentally safe.

# 4 Management Options: Managing Properly Cow-Calf Operations to Improve Water Quality and Pasture Sustainability

### 4.1 Improving Pasture Sustainability

The cow-calf (Bos taurus) industry in subtropical United States and other parts of the world depends almost totally on grazed pasture areas. Thus, the establishment of complete, uniform stand of bahiagrass in a short time period is economically vital. Failure to obtain a high-quality bahiagrass stand early means the loss of not only the initial investment costs, but also both production and its cash value. Forage production often requires significant inputs of lime, N fertilizer and less frequently of P and K fertilizers. Domestic wastewater sludge or sewage sludge, composted urban plant debris, waste lime, phosphogypsum and dredged materials are examples of materials that can be used for fertilizing and liming pastures. Beef cattle producers throughout the United States need better forage management systems to reduce input costs and protect environmental quality.

The ability to reuse lake-dredged and domestic sewage sludge materials for agricultural purposes is important because it reduces the need for offshore disposal and provides an alternative to disposal of these materials in landfills that are already overtaxed. Often these materials can be obtained at little or no cost to the farmers or landowners. Thus, forage production offers an alternative to waste management since nutrients in the lake-dredged materials and biosolids are recycled into crops that are not directly consumed by humans. Results have shown the favorable influence that biosolids and lake-dredged materials had on bahiagrass during its early establishment in sandy subtropical beef cattle pasture areas in south central Florida. Some of the promising effects of added biosolids and lake-dredged materials on soil quality and on early establishment of bahiagrass are summarized below.

### 4.1.1 Biosolids as Nutrient Source

Biosolids usually are applied at agronomic rates designed to supply crops with adequate N nutrition. Biosolids contain a substantial amount of N (typically 3 to 6 % by weight). The N is not immediately available to crops, but is released slowly by biological activity. Since biosolids are produce and handled by different processes at different treatment plants, it is important to know if those treatment processes affect how much N becomes available to plants. Nutrients in municipal residuals produced annually in the United States account for about 2.5% of the total N, 6% of the P, and 0.5% of the K applied on farms each year (Muse et al., 1991).

The field experiment was conducted at the University of Florida Agricultural Research and Education Center, Ona, FL (27°26'N, 82°55'W) on a Pomona fine sandy soil. With the exception of the control, bahiagrass plots received annual biosolids and chemical fertilizers applications to supply 90 or 180 kg total

With sewage sludge

N ha<sup>-1</sup> yr<sup>-1</sup> from 1997 to 2000. Land application of biosolids and fertilizer ceased in 2001 season. In early April 1998, 1999, and 2000, plots were mowed to 5-cm stubble and treated with the respective N source amendments. The experimental design was three randomized complete blocks with nine N-source treatments: ammonium nitrate (AMN), slurry biosolids of pH 7 (SBS7), slurry biosolids of pH 11 (SBS11), limestabilized cake biosolids (CBS), each applied to supply 90 or 180 kg N ha<sup>-1</sup>, and a nonfertilized control (Control). Application rates of biosolids were calculated based on the concentration of total solids in materials as determined by the American Public Health Association SM 2540G methods (APHA, 1989) and N in solids. The actual amount of biosolids applications was based on the amount required to supply 90 and 180 kg N ha<sup>-1</sup>. Sewage sludge materials were weighed in buckets and uniformly applied to respective bahiagrass plots.

Forage yield of bahiagrass was significantly ( $P \leq 0.001$ ) affected by the different biosolids in all years (1998 to 2002), but not by the interaction effects of year × treatments. Although yield trend was declining from 1988 to 2002, forage yield of bahiagrass that received biosolids were consistently and significantly ( $P \leq 0.05$ ) greater than the forage yield of the unfertilized bahiagrass (Table 5). The bahiagrass fertilized with SBS11-180 had the greatest forage yield in 1998 ( $5.1 \pm 0.4$  Mg ha<sup>-1</sup>), 1999 ( $4.6 \pm 0.2$  Mg ha<sup>-1</sup>), 2000 ( $4.5 \pm 0.2$  Mg ha<sup>-1</sup>), and in 2002 ( $3.3 \pm 0.6$  Mg ha<sup>-1</sup>). Forage yield of bahiagrass fertilized with AMN-180 was significantly greater than

Without sewage sludge

**Table 5** Comparison on forage yield (Mg ha<sup>-1</sup>; mean  $\pm$  S.D.) of bahiagrass among years with repeated application of biosolids (1998, 1999, and 2000) and with no biosolids application (2002) (*Source: Sigua, 2008*)

Nitrogen sources*	1998	1999	2000	2002	
Control	$2.4 \pm 0.5 d^{**}$	$1.8 \pm 0.2c$	$1.4 \pm 0.3$ d	$1.2 \pm 0.2c$	
AMN-90	$4.3 \pm 0.2$ ab	$3.7 \pm 0.1b$	$2.1 \pm 0.1$ cd	$2.1 \pm 0.3 \text{bc}$	
AMN-180	$4.7 \pm 0.4a$	$4.7 \pm 0.02a$	$3.2 \pm 0.3b$	$2.2 \pm 0.4 \mathrm{b}$	
SBS7-90	$4.4 \pm 0.4$ ab	$3.1 \pm 0.3b$	$2.2 \pm 0.4$ bcd	$2.5 \pm 0.5 ab$	
SBS7-180	$5.0 \pm 0.5$ a	$5.1 \pm 0.2a$	$2.6 \pm 0.2 \mathrm{bc}$	$2.3 \pm 0.5 \mathrm{b}$	
SBS11-90	$4.1 \pm 0.5$ abc	$3.3 \pm 0.3b$	$1.9 \pm 0.3$ cd	$1.9 \pm 0.2 bc$	
SBS11-180	$5.1 \pm 0.4a$	$4.6 \pm 0.2a$	$4.5 \pm 0.2a$	$3.3 \pm 0.6a$	
CBS-90	$2.9 \pm 0.4$ cd	$2.2 \pm 0.2c$	$1.8 \pm 0.6$ cd	$2.5\pm0.5ab$	
CBS-180	$3.3 \pm 0.3$ bcd	$3.3 \pm 0.1b$	$2.7 \pm 0.2 bc$	$2.5\pm0.5ab$	
*			1. 1.1 6 11 11 6		

\*AMN – ammonium nitrate; SBS7 – slurry biosolids of pH 7; SBS11 – slurry biosolids of pH 11; CBS7 – limed-stabilized cake biosolids; 90–90 kg N ha<sup>-1</sup>; 180–180 kg N ha<sup>-1</sup>.

\*\*Mean values in each column followed by the same letter(s) are not different (P > 0.05) according to the Duncan's multiple range test.

those of the unfertilized bahiagrass in 1998 and 1999, but not in 2000 and in 2002. Although SBS11-180 had the greatest residual effect (170%) in 2002, CBS-90 and CBS-180 had more pronounced effects when compared with the other sewage sludge sources because their relative impact on forage yield compared with the control between years with (1997–2000) and without (2002) sewage sludge applications increased from 30% to 110% and 70% to 110%, or net increases of 267% and 57% in forage yield change, respectively.

The residual effects of applied sewage sludge on bahiagrass yield expressed as percent forage yield change over the unfertilized bahiagrass are shown in Figure 8. Residual effects of AMN-90 (-6%), AMN-180 (-31%), SBS7-80 (-21%), and SBS7-180 (-17%) declined (negative) with time, but the residual effects of applied SBS11-180 (+13%), CBS-90 (+267%), and CBS-180 (+57%) were positive over time although sewage sludge application ceased after harvest in 2000. The percent forage yield change of bahiagrass fertilized with SBS11-180, CBS-90, and CBS-180 during years when sewage sludges were applied (1998–2000) were 150%, 30%, and 70% compared with percent forage yield change of 170%, 110%, and 110% in 2002 (when sewage sludge applications ceased), respectively.

The residual effects on forage yield of applied CBS-90 (+267%) and CBS-180 (+57%) relative to the control increased with time although biosolids applications ceased after the 2000 harvest season. This was probably due to the higher concentration of organic N in addition to the liming property of CBS. Liming the field could have some direct and indirect effects on forage productivity and on the nutrient status of the soils. Perhaps the single direct benefit of liming is the reduction in acidity and solubility of aluminum and manganese (Peevy et al., 1972). Some of the indirect benefits of liming pasture fields among others would include: enhancing P and microelement availability, nitrification, N fixation, and improving soil physical conditions (Nelson, 1980; Tisdale and Nelson, 1975; Russell, 1973). Dried and composted biosolids have slower rates of N release and in case of CBS with much higher solids concentration (500 000 mg  $L^{-1}$ ); more N will be released in the second, third, or even the fifth year after the initial application due to higher amount of organic N than ammonium N. The proportions of ammonium and organic N in biosolids vary with the stabilization process.

### 4.1.2 Lake-Dredged Materials as Nutrient Source

The field study was adjacent to the Coleman Landing spoil disposal site in Sumter County, FL. Each plot (961 m<sup>2</sup>) was excavated to a depth of about 28 cm, and existing natural soil (NS) and organic materials were completely removed. Excavated NS materials were placed at the south end of the test plots. Existing vegetation from each plot was totally removed prior to backfilling each plot with different ratios of NS and lake-dredge materials (LDM): (100% NS + 0% LDM); (75% NS + 25% LDM);(50% NS + 50% LDM); (25% NS + 75% LDM);and (0% NS + 100% LDM). These ratios of NS to LDM represent the treatment combinations of LDM0; LDM25; LDM50; LDM75; and LDM100, respectively. Natural soils that were excavated were backfilled to each plot along with lake-dredged materials that were hauled from the adjacent settling pond. The total amount of lake-dredged materials and natural soils that was placed back on each test plot was in accordance with the different ratios of lake-dredged materials and natural soils that were described above. After mixing the natural soils and lake-dredged materials, each of the test plots was disked to a uniform depth of 28 cm. Plots were disked in an alternate direction until lake-dredged materials and natural soils were uniformly mixed. Each plot was seeded with bahiagrass at a rate of 6 kg plot<sup>-1</sup>, followed by dragging a section of chain link fence across each test plot to ensure that bahiagrass seeds were in good contact with the natural soils and lake-dredged materials. Field layout was based on the principle of a completely randomized block design with four replications.

The forage yield of bahiagrass at 112, 238, and 546 Julian days after seeding are shown in Fig. 10. Forage yield of bahiagrass varied significantly ( $P \le 0.001$ ) among plots with lake-dredged materials additions. The greatest forage yield of  $673 \pm 233$  kg ha<sup>-1</sup> at Julian day 112 was from plots amended with 50% lakedredged materials while bahiagrass in plots amended with 100% lake-dredged materials and 75% lakedredged materials had the highest forage yield at Julian days 238 and 546 with average forage yield of 3349  $\pm$  174 and 4109  $\pm$  220 kg ha<sup>-1</sup>, respectively (Fig. 10). The lowest forage yield of 89  $\pm$  63, 1513  $\pm$  166, and 1263  $\pm$  116 kg ha<sup>-1</sup> were from the control plots for Julian days 112, 238, and 546, respectively (Fig. 10). The average forage yield increase of



**Fig. 10** Forage yield of bahiagrass (Julian days 112 to 546) as affected by varying levels of dredged materials application. Forage yield from plots with or without lake-dredged materials

bahiagrass in plots amended with lake-dredged materials (averaged across treatments) was 512%, 82%, and 173% when compared with bahiagrass in control plots with 0% lake-dredged materials for Julian days 112, 238, and 546, respectively (Fig. 10). These data show the favorable influence that lake-dredged materials had on forage yield of bahiagrass during its early establishment in subtropical beef cattle pastures.

Mean forage yield of bahiagrass during Julian day 112 in plots with 50% lake-dredged materials of 673  $\pm$  233 kg ha<sup>-1</sup> was not significantly different from that in plots with 75% lake-dredged materials (654  $\pm$  106 kg ha<sup>-1</sup>), but was greater than that in plots with 25% lake-dredged materials (378  $\pm$  185 kg ha<sup>-1</sup>) and 0% lake-dredged materials (Fig. 10). For Julian day 238, the greatest forage yield among plots amended with lake-dredged materials was from plots with 100% lakedredged materials (3349  $\pm$  174 kg ha<sup>-1</sup>). The lowest forage yield of 1513  $\pm$  166 kg ha<sup>-1</sup> was from plots with 0% lake-dredged materials. Mean forage yield of bahiagrass in plots with 50% lake-dredged materials of 2467  $\pm$  320 kg ha<sup>-1</sup> was not significantly different from that in plots with 75% lake-dredged materials  $(2467 \pm 320 \text{ kg ha}^{-1})$  and 25% lake-dredged materials  $(2409 \pm 423 \text{ kg ha}^{-1})$ , but was greater than that in plots with 0% lake-dredged materials (Fig. 10).

# are significantly different ( $P \leq 0.05$ ) at Julian days 112, 238, and 546 when superscripts located at top of bars are different (*Source: Sigua, 2008*)

### 4.1.3 Using Legumes to Enhance Nitrogen Fertility in Bahiagrass Pastures

Beef cattle pastures in the United States and other parts of the world are typically dominated by tropical grasses such as bahiagrass or bermudagrass; drawbacks to these grasses include limited growth during the winter, relatively low nutritional value, and high N fertilizer costs. The development of effective grass-legume pastures for beef cattle production would improve nutritional value of the pastures, minimize N fertilizer input, and better manage nutrient cycling to enhance water quality. Improving the nutritional value of these grass pastures by the addition of legume will improve beef cattle gains in this region and impact both quantity and cost of the red meat supply in the United States and other parts of the world.

The establishment and maintenance of persistent grass-legume pastures is a key option to increase productivity and profitability of beef cattle production systems. Despite the great potential to improve sustainability of animal production, the adoption of technologies has been limited and slow. The integration of legumes (*Arachis glabrata*, Benth) and non-legumes (bahiagrass, *Paspalum notatum*, Flugge) can produce synergistic effects that can minimize external inputs, particularly N fertilizers in agricultural ecosystems. There is a need to carry out research that will help to understand the different factors affecting the fixation and release of biologically fixed N to non-legume crops. The integration of legumes and non-legumes (grass) can produce synergistic effects that can minimize external inputs, particularly N fertilizers in agricultural ecosystems (Saikia and Jain, 2007; Rochester et al., 2001; Omay et al., 1998; Keisling et al., 1994; Oyer and Touchton, 1990).

Symbiotic  $N_2$  fixation may provide N for the growth of legumes and eventually to intercropped plants, such as grasses through the transfer of N from the legumes. An increased understanding of the mechanisms of N transfer in grass-legume mixtures will help increase forage productivity. Nitrogen transfer if properly documented and quantified can lead to improved management systems and increased productivity of grasslegume mixtures (e.g., bahiagrass-peanut mixtures), while minimizing the use of inorganic N fertilizers. This will increase profits and decrease the impact of agriculture on the environment.

### 4.1.4 Managing Nutrients Across Paddocks to Improve Water Quality

#### Nutrient Balance

Nutrient balance in the ecosystem involves profitability of the agricultural enterprise and commitments to resource management to maintain quality of air, water and land resources. The role of nutrient management in livestock systems takes on new meaning as producers and the public together consider economic and noneconomic issues (Nelson, 1999). The intensification of livestock production with its associated increased demand for fodder has encouraged farmers to rely more heavily on chemical fertilizers and imported feeds, and very often the waste is considered as a disposal problem rather than useful source of plant nutrients (Hooda et al., 2000). It should be noted that for a farm to be sustainable, its P or N budget should balance, at least after soil reserves are brought up to desired levels for sustainable production. If there is a net loss of P or N, the farm's soils will eventually become depleted and if there is an excess, the likelihood of pollution is greater (Van Horn et al., 1996). Effective use and cycling of P or N is critical for pasture pro-



**Fig. 11** Generalized schematic showing different phosphorus compartments in forage-based pasture system (*modified from Sigua et al., 2006c*) (+ Loss or export from the system)

ductivity and environmental stability. Phosphorus or N (not shown) cycling in pastures is complex and interrelated and pasture management practices influence the interactions and transformations occurring within the P cycle (Fig. 11).

Increased loss of nutrients in agricultural runoff has potentially serious ecological and public health implications (Hooda et al., 2000). Nitrogen and P are particularly important as both are implicated in aquatic eutrophication (Levine and Schindler, 1989). Any approach that controls N and/or P losses from agriculture to water must begin with the long-term objective of increasing N and/or P efficiency by attempting to balance N or P inputs with N or P outputs within a watershed. Reducing N or P loss in agricultural runoff may be brought about by Best Management Practices (BMPs) that control the source and transport of N and/or P (Table 6).

A review paper published by Shigaki et al. (2006) on animal-based agriculture options for the future contained excellent discussion on the source and transport management of nutrients in the watershed. Source management attempts to minimize the buildup in the soil above levels sufficient for optimum crop growth while transport management refers to efforts to control the movement of nutrients from soils to sensitive locations such as bodies of fresh water. As shown in Table 6, there are several measures available to minimize the potential for nutrient losses in agricultural runoff, which address sources and transport of P and/or N. Important measures to be considered are those that attempt to decrease the surplus of nutrients (P or N) **Table 6** Modified best management practices for the control of diffuse sources of agricultural nutrients (*Source: Shigaki et al., 2006*)

- I. Source BMPs Practices that minimize nutrients loss at the origin
  - 1. Attempts to match animal requirements for N or P with feeds P or N
  - 2. Enzyme added to feeds to increase nutrient utilization by animals
  - 3. Test soils and manures to optimize N and P management
  - 4. Chemically treat manure to reduce P solubility (e.g., alum, fly ash, etc.)
  - 5. Biologically treat manure (e.g., microbial enhancement)
  - 6. Calibrate fertilizer and manure application equipment
  - 7. Apply proper amount of fertilizer
  - 8. Careful timing of fertilizer application to avoid imminent heavy rainfalls

II. Transport BMPs - Practices that minimize transport of nutrients

- 1. Minimize erosion, runoff and leaching of nutrients
- 2. Use cover crops to protect soil surface from erosion
- 3. Install filter strips and other conservation buffers to trap eroded P
- 4. Manage riparian zones, grassed waterways and wetlands to trap eroded nutrients
- 5. Stream bank fencing to exclude animal from water course

III. Source and transport BMPs - Systems approach to minimize nutrient losses

- 1. Retain crop residues and reduce tillage to minimize erosion and runoff
- 2. Proper grazing management to minimize erosion and runoff
- 3. Install and maintain manure handling systems
- 4. Implement nutrient management plan for the farms

in localized areas. Shigaki et al. (2006) suggested the following measures: (1) dietary P reduction; (2) feed additives that enhance P utilization by animals; (3) alternative uses for manure other than land application; and (4) transporting manure to P- or N-deficient areas.

# Water Quality Best Management Practices (BMP) for Florida Cow-Calf Operations

Although cow-calf operations in Florida are generally low-intensity agriculture with relatively low levels of pollutant discharged off-site, certain sites and management practices may contribute to violations of State water quality standards. Under these situations, cattle ranches may contribute elevated levels of P, N, sediments, bacteria and oxygen-demanding organic materials. The potential for discharges from cow-calf operations to cause water quality violations varies greatly, depending on soil type, slope, drainage features, stocking rate, nutrient management and other factors. In general, areas where cattle tend to congregate or have access to water bodies may have the greatest potential to contribute to water pollution (Sigua and Coleman, 2007). Spearheaded by the Florida Cattlemen's Association (FCA) and drawn up in a unique partnership between producers and regulators, the BMP manual serves as both roadmap and vehicle to enhance and protect water quality in Florida.

In 1997, the Florida Cattlemen Association began drafting a common sense, economically-viable guidelines for production practices designed to protect water bodies and maintain compliance with the State of Florida water quality standards. The manual describes the water quality best management practices (BMPs) for beef cow-calf operations in Florida. The different practices are specifically targeted for beef cowcalf operations in Florida and the activities that normally occur in conjunction with beef cattle production. These are not rules or regulations, but voluntary best management practices for Florida ranches that consider good water quality conditions.

The manual is heralded as a unique consensus document that outlines common sense, economically and technically feasible production and management practices that enhance and protect Florida's water resources. It is designed specifically for Florida's cowcalf operations; it does not apply to concentrated animal feeding operations (CAFOs), which generally require a permit (see the 107-page BMP manual at http://jefferson.ifas.ufl.edu/old/agpages/cow\_ calf\_bmp/BMPManual.pdf). Although the manual was designed for statewide use in Florida, the general premise and principles can be adopted elsewhere.

### 5 Summary and Conclusions

Current pasture management including cattle rotation in terms of grazing days and current fertilizer (inorganic + manures + urine) application rates for bahiagrass pastures in subtropical regions of USA offer little potential for negatively impacting the environment. Properly managed livestock operations contribute negligible loads of total P and N to shallow groundwater and surface water. Overall, there was no buildup of soil total P and N in bahiagrass-based pasture. These observations may help to renew the focus on improving fertilizer efficiency in subtropical beef cattle systems, and maintaining a balance of P and/or N removed to P and/or N added to ensure healthy forage growth and minimize P or N runoff.

Contrary to early perception, forage-based animal production systems with grazing are not likely one of the major sources of non-point source P pollution that are contributing to the degradation of water quality in lakes, reservoirs, rivers, and ground water aquifers, but perennially grass-covered pastures are associated with a number of environmental benefits. Continuous grass cover leads to the accumulation of soil organic matter, sequestering carbon in the soil and thereby reducing the potential  $CO_2$  accumulation in the atmosphere. The increase in soil organic matter is also related to soil quality, with improvements in soil structure, aeration and microbial activity. Effective use and cycling of N or P is critical for pasture productivity and environmental stability. In addition to speeding up N or P recycling from the grass, grazing animals can also increase N or P losses in the system by increasing leaching potential due to concentrating N into small volumes of soil under dung and urine patches, redistributing N or P around the landscape, and removal of N or P in the form of animal products.

The overall goal efforts to reduce N or P losses from animal-based agriculture should be to balance off-farm P inputs in feed and fertilizer with outputs to the environment. Source and transport control strategies can provide the basis to increase N and P efficiency in agricultural systems.

Overall,

Forage-based animal production systems as suggested by regulators are not the major sources of non-point source nutrients pollution that are contributing to the degradation of water quality in lakes, reservoirs, rivers, and ground water aquifers; and

• Properly managed cow-calf operations in subtropical agro-ecosystem would not likely be the major contributors to excess loads of N or P in surface water and/or shallow groundwater.

### References

- Allen L.H. Jr., Ruddell J.M., Rutter G.J., Yates P. (1982) Land effects of Taylor Creek water quality, in: Kruse E.G. et al. (Eds.), Environmentally Sound Water and Soil Management, Amer. Soc. Civil Eng., New York, pp. 67–77.
- Allen L.H. Jr., Stewart E.H., Knisel W.G. Jr., Stack R.A. (1976) Seasonal variation in runoff and water quality from the Taylor Creek watershed, Okeechobee County, FL, Soil Crop Sci. Soc. Fla. Proc. 35, 126–138.
- American Public Health Association (1992) Standard methods for the examination of water and wastewater, 18th ed., APHA, Washington.
- American Public Health Association (1989) Standard Methods for the examination of water and wastewater, 17th ed. APHA, Washington, DC.
- Anderson D.W., Coleman D.C. (1985) The dynamics of organic matter in grassland soils, J. Soil Water Conserv. 40, 211–215.
- Andrew M.H., Lange R.T. (1986) Development of a new piosphere in arid chenopod shrubland grazed by sheep. 1) Changes to the soil surface, Aust. J. Ecol. 3, 336–339.
- Arthington J., Bohlen P., Roka F. (2003) Effect of stocking rate on measures of cow-calf productivity and nutrient load in surface water runoff, University of Florida IFAS Extension, Gainesville, FL, 4 p.
- Asmussen L.E., Sheridan J.M., Allison H.D. (1975) Water quality inventory of the Southern Coastal Plain and Atlantic Coastal Flatwoods of Georgia, USDA-ARS Publ. ARS-S-49, U.S. Government Printing Office, Washington, DC.
- Bogges C.F., Flaig E.G., Fluck R.C. (1995) Phosphorous budgetbasin relationships for Lake Okeechobee tributary basins, Ecol. Eng. 5, 143–162.
- Botcher A.B., Tremwel T.K., Campbell K.L. (1999) Phosphorus management in flatwood (Spodosols) soils, in: Reddy K.R. et al. (Eds.), Phosphorus Biogeochemistry in Subtropical Ecosystems, Lewis Publishers, Boca Raton, FL, pp. 405–423.
- Bowers E.J., Hammond A.C., Chase C.C. Jr., Olson T.A. (1995) Effect of breed on indicators of heat tolerance and grazing activity in lactating Angus and Brahman cows in Florida, J. Anim. Sci. 73 (Suppl. 1), 131.
- Brezonik P.L. (1984) Trophic state indices: rational for multivariate approaches, Lake and Reservoir Management, EPA 440/5-84-001, USEPA, Washington, DC, pp. 441–445.
- Briske D.D., Richards J.H. (1995) Plant responses to defoliation: a physiological, morphological, and demographic evaluation, in: Bedunah D.J., Sonebee R.E. (Eds.), Wildland Plants: Physiological Ecology and Developmental Morphology, pp. 635–710.
- Brooks K.N., Folliot P.N., Gregersen H.M., DeBano L.F. (1997) Hydrology and the management of watersheds, 2nd ed., Iowa State University Press, Ames, IA.

- Carlson R.E. (1977) A trophic state index for lakes, Limnol. Oceanogr. 22, 361–369.
- Carlson R.E. (1983) Discussion on "Using differences among Carlson's trophic state index values in regional water quality assessment", Water Resour. Bull. 19, 307–309.
- Chambliss C.G. (1999) Florida Forage Handbook, Univ. Florida Coop. Ext. Serv. SP 253.
- Chambliss C.G., Miller P., Lord E. (2001) Florida cow-calf management, 2nd Edition-Forages, University of Florida, IFAS Extension, p. 12.
- Chiandini G., Vighi M. (1974) The N:P ratio and tests with Selenasrum to predict eutropication in lakes, Water Res. 8, 1063–1069.
- Coale F.J., Olear J.A. (1996) The relationship between soil test phosphorus level and the concentration of dissolved and potentially transportable phosphorus in field drainage water, Technical Report, Chesapeake Research Consortium, Inc., Edgewater, Maryland.
- Decau M.L., Simon J.C., Jacquet A. (2003) Fate of urine in three soils throughout a grazing season, J. Environ Qual. 32, 1405–1413.
- Dormaar J.F. (1992) Decomposition as a process in natural grasslands, in: Coupland R.T. (Ed.), Ecosystems of the World, Elsevier, New York, pp. 121–136.
- Edwards D.R., Hutchens T.K., Rhodes R.W., Larson B.T., Dunn L. (2000) Quality of runoff from plots with simulated grazing, J. Am. Water Resour. Assoc. 36, 1063–1073.
- Fleischner T.L. (1994) Ecological costs of livestock grazing in western North America, Conserv. Biol. 8, 629–644.
- Franzluebbers A.F., Stuedemann J.A., Schomberg H.H. (2000) Spatial distribution of soil carbon and nitrogen pools under grazed tall fescue, Soil Sci. Soc. Am. J. 64, 635–639.
- Ganskopp D. (2001) Manipulating cattle distribution with salt and water in large arid-land pastures: a GPS/GIS assessment, Appl. Anim. Behav. Sci. 73, 251–262.
- Gary H.L., Johnson S.R., Ponce S.L. (1983) Cattle grazing impact on surface water quality in a Colorado Front Range Stream, J. Soil Water Conserv. 38, 124–128.
- Gburek W.J., Sharpley A.N. (1998) Hydrologic controls on phosphorus loss from upland agricultural watersheds, J. Environ. Qual. 27, 267–277.
- Hammond A.C., Olson T.A. (1994) Rectal temperature and grazing time in selected beef cattle breeds under tropical summer conditions in subtropical Florida, Trop. Agric. (Trinidad) 71, 128–134.
- Haynes R.J. (1981) Competitive aspects of the grass-legume association, Adv. Agron. 33, 227–261.
- Haynes R.J., Williams P.H. (1993) Nutrient cycling and soil fertility in grazed pasture ecosystem, Adv. Agron. 49, 119–199.
- Herbel C.H., Ares F.N., Nelson A.B. (1967) Grazing distribution patterns of Hereford and Santa Gertrudis cattle on southern New Mexico range, J. Range Manage. 20, 296–298.
- Holechek J.L. (1988) An approach for setting stocking rate, Rangeland 10, 10–14.
- Holt J.A., Coventry R.J. (1990) Nutrient cycling in Australian savannas, J. Biogeogr. 17, 427–432.
- Hooda P.S., Edwards A.C., Anderson H.A., Miller A. (2000) A review of water quality concerns in livestock farming areas, Sci. Total Environ. 250, 143–167.

- Hooda P.S., Rendell A.R., Edwards A.C., Withers P.J., Aitken M.N., Truesdale V.W. (2000) Relating soil phosphorus indices to potential phosphorus release to water, J. Environ. Qual. 29, 1166–1171.
- Hubbard R.K., Sheridan J.M. (1983) Water and nitrate-nitrogen losses from a small, upland, coastal plain watershed, J. Environ. Qual. 12, 291–295.
- Janzen H.H., Campbell C.A., Ellert B.H., Bremer E. (1997) Soil organic matter dynamics and their relation to soil quality, in: Gregorich E.G., Carter M.R. (Eds.), Soil Quality for Crop Production and Ecosystem Health, Elsevier Science Publ. Amsterdam, The Netherlands, pp. 168–185.
- Janzen H.H., Campbell C.A., Gregorich E.G., Ellert B.H. (1998) Soil carbon dynamics in Canadian Agroecosystems, in: Lal R. et al. (Eds.), Soil Processes and the Carbon Cycle, CRC Press, Boca Raton, FL, pp. 57–81.
- Johnson J., Eckert D. (1995) Best management practices: land application of animal manure, The Ohio State University Extension Agronomy Facts, AGF-208-95, Columbus, OH, 10 p.
- Kauffman J.B., Krueger W.C. (1984) Livestock impacts on riparian ecosystems and streamside management implications: a review, J. Range Manage. 37, 430–437.
- Keisling T.C., Scott H.D., Waddle B.A., Williams W., Frans R.E. (1994) Winter cover crops influence on cotton yield and selected soil properties, Commun. Soil Sci. Plant Anal. 25, 3087–3100.
- Kellogg R.L., Lander C.H., Moffit D.C., Gollehon N. (2000) Manure nutrients relative to the capacity of cropland and pastureland to assimilate nutrients: Spatial and temporal trends for the United States, USDA-NRCS, Washington, DC, p. 93.
- Khaleel R., Reddy K.R., Overcash M.R. (1980) Transport of potential pollutants in runoff water from land areas receiving animal wastes: A review, Water Res. 14, 421–426.
- Kieft T.L. (1994) Grazing and plant-canopy effects on semiarid soil microbial biomass and respiration, Biol. Fertil. Soils 18, 155–162.
- Kieft T.L., White C.S., Loftin S.R. Aguilar R., Craig J.A., Skaar D.A. (1998) Temporal dynamics in soil carbon and nitrogen resources at a grassland-shrubland ecotone, Ecology 79, 671–683.
- Klemmendson J.O., Tiedemann A.R. (1995) Effects of nutrient stress, in: Bedunah D.J., Sosebee R. (Eds.), Wildland Plants: Physiological Ecology and Developmental Morphology, Society of Range Management, Denver, CO, pp. 414–439.
- Kotak B.G., Kenefick S.L., Fritz D.L., Rousseaux C.G., Prepas E.E., Hrudey S.E. (1993) Occurrences and toxicological evaluation of cyanobacterial toxins in Alberta lakes and farm dugouts, Water Res. 27, 495–506.
- Lauenroth W.K., Milchunas D.G., Dodd J.L., Hart R.H., Heitschmidt R.K., Rittenhouse L.R. (1994) Effects of grazing on ecosystems of the Great Plains, in: Vavra M., Laycock W.A., Pieper R.D. (Eds.), Ecological implications of livestock herbivory in the West. Society for Range Management, Denver, CO.
- Levine S.L., Schindler D.W. (1989) Phosphorus, nitrogen and carbon dynamics of experimental lake 303 during recovery from eutrophication, Can. J. Fish. Aquat. Sci. 46, 2–10.
- Martin S.C., Ward D.E. (1973) Salt and meal-salt help distribute cattle use on semi-desert range, J. Range Manage. 26, 94–97.

- Mathews B.W., Sollenberger L.E., Nair V.D., Staples C.R. (1994) Impact of grazing management on soil nitrogen, phosphorus, potassium, and sulfur distribution, J. Environ. Qual. 23, 1006–1013.
- Mathews B.W., Tritschler J.P., Carpenter J.R., Sollenberger L.E. (1999) Soil macronutrients distribution in rotationally stocked kikuyugrass paddocks with short and long grazing periods, Commun. Soil Sci. Plant Anal. 30, 557–571.
- Milchunas D.G., Lauenroth W.K. (1993) Quantitative effects of grazing on vegetation and soils over a global range of environments, Ecol. Monogr. 63, 327–366.
- Muse J.K., Mitchell C.C., Mullins G.L. (1991) Land application of sludge, Ala. Coop. Ext. Serv. Circ. ANR-607, Auburn University, Auburn, AL.
- Nelson C.J. (1999) Managing nutrients across regions of the United States, J. Anim. Sci. 77, 90–100.
- Nelson W.L. (1980) Agricultural liming: its effect on soil fertility, plant nutrition, and yields, Proc. 1st Natl. Conf. Agric. Limestone, Nashville, TN, pp. 34–39.
- Omay A.B., Rice C.W., Maddux L.D., Gordon W.B. (1998) Corn yield and nitrogen uptake in monoculture and in rotation with soybean, Soil Sci. Soc. Am. J. 62, 1596–1603.
- Oyer L.J., Touchton J.T. (1990) Utilizing legume cropping systems to reduce nitrogen fertilizer requirements to conservation-tilled corn, Agron. J. 82, 1123–1127.
- Palmstrom N.S., Carlson R.E, Cooke G.D. (1988) Potential links between eutrophication and formation of carcinogens in drinking water, Lake and Reservoir Management 4, 1–15.
- Peevy W.J., Brupbacher R.H., Sedberry J.E. Jr. (1972) Effects of some liming materials and calcium sulfate on soil reaction and exchangeable calcium and magnesium, LSU Agr. Center, Baton Rouge, LA, p. 8.
- Rochester I.J., Peoples M.B., Hulugalle N.R., Gault R.R., Constable G.A. (2001) Using legumes to enhance nitrogen fertility and improve soil condition in cotton cropping systems, Field Crops Res. 70, 27–41.
- Romkens J.J.M., Nelson W., Mannering J.V. (1973) Nitrogen and phosphorus composition of surface runoff as affected by tillage method, J. Environ. Qual. 2, 292–295.
- Ruess R.W., McNaughton S.J. (1987) Grazing and the dynamics of nutrient and energy regulated microbial processes in the Serengeti grasslands, Oikos 49, 101–110.
- Russell W. E. (1973) Soil conditions and plant growth, William Clowes & Sons, Limited, London, p. 849.
- Saikia S.P., Jain V. (2007) Biological nitrogen fixation with nonlegumes: an achievable target or a dogma, Curr. Sci. 92, 317–321.
- Sakamoto M. (1966) Primary production by phytoplankton community in some Japanese lakes and its dependence on lake depth, Arch. Hydrobiol. 62, 1–28.
- Schindler D.W. (1974) Eutrophication and recovery in experimental lakes: Implications for lake management, Science 184, 260–262.
- Schmidt K.P., Sturgul S.J. (1989) Nutrient and pesticide best management practices for Wisconsin farms, WDAPTCP Technical Bulletin ARM-1, University of Wisconsin and Wisconsin Department of Agriculture, Madison, WI.
- Senft R.L., Rittenhouse L.R., Woodmanse R.G. (1983) The use of regression models to predict spatial patterns of cattle behavior, J. Range Manage. 36, 553–557.

- Senft R.L., Rittenhouse L.R., Woodmanse R.G. (1985) Factors influencing patterns of cattle grazing behavior on shortgrass steppe, J. Range Manage. 38, 82–87.
- Sharpley A.N. (1997) Dispelling common myths about phosphorus in agriculture and the environment. Technical Report, Watershed Science Institute, USDA Natural Resources Conservation Service, Aiken Center, University of Vermont, Burlington, Vermont.
- Sharpley A.N., Withers P.J.A. (1994) The environmentally sound management of agricultural phosphorus, Fertil. Res. 39, 133–146.
- Sharpley A.N., Daniel T.C., Sims J.T., Pote D.H. (1996) Determining environmentally sound soil phosphorus levels, J. Soil Water Conserv. 51, 160–166.
- Shigaki F., Sharpley A., Prochnow L.I. (2006) Animal-based agriculture, phosphorus management and water quality in Brazil: Options for the Future, Sci. Agric. 63, 194–209.
- Sigua G.C. (2004) Impact of cattle congregation sites on soil nutrients and soil compaction, USDA-ARS-STARS Field Day Manual, Brooksville, FL, pp. 10–15.
- Sigua G.C. (2008) Recycling biosolids and lake-dredged materials to pasture-based animal agriculture: alternative nutrient sources for forage productivity and sustainability, Agron. Sustain. Dev. 29, 143–160.
- Sigua G.C., Steward J.S. (2000) Establishing pollutant load reduction targets for the Indian River Lagoon, Florida, J. Am. Water Resour. Assoc. 36, 123–132.
- Sigua G.C., Tweedale W.A. (2003) Watershed scale assessment of nitrogen and phosphorus loadings in the Indian River Lagoon Basin, FL, J. Environ. Manage. 67, 361–370.
- Sigua G.C., Coleman S.W., Williams M. (2005) Cattle congregation sites do not appear to impact soil nutrients buildup, 2005, The Florida Cattleman and Livestock Journal 69, 76–79.
- Sigua G.C., Hubbard R., Coleman S.W. (2009a) Nitrogen in soils, plants and shallow groundwater beneath a bahiagrassbased pasture with cow-calf operations, Nutr. Cycl. Agroecosyst. J., DOI: 10.1007/s10705-009-9281-8.
- Sigua G.C., Hubbard R.K., Coleman S.W., Williams M. (2009b) Nitrogen in soils, surface water and shallow groundwater in a bahiagrass pasture of southern Florida, USA, Nutr. Cycl. Agroecosyst. J., DOI: 10.1007/s10705-009-928-8.
- Sigua G.C., Kang W.J., Coleman S.W. (2006a) Soil profile distribution of phosphorus and other soil nutrients following wetland conversion to beef cattle pasture, J. Environ. Qual. 35, 2374–2382.
- Sigua G.C., Williams M.J., Coleman S.W. (2006b) Long-term effects of grazing and haying on soil nutrient dynamics in forage-based beef cattle operations, J. Sustain. Agr. 29, 115–134.
- Sigua G.C., Williams M.J., Coleman S.W., Starks R. (2006c) Nitrogen and phosphorus status of soils and trophic state of lakes associated with forage-based beef cattle operations in Florida, J. Environ. Qual. 35, 240–252.
- Sigua G.C., Steward J.S., Tweedale W.A. (2000) Water quality monitoring and biological integrity assessment in the Indian River Lagoon, Florida I. Status and trends (1988–1994), J. Environ. Manage. 25, 199–209.
- Sigua G.C., Williams M.J., Coleman S.W. (2004) Levels and changes of soil phosphorus in the subtropical beef cattle pastures, Commun. Soil Sci. Plant Anal. 35, 975–990.

- Stout W.L., Fales S.S., Muller L.D., Schnabel R.R., Priddy W.E., Elwinger G.F. (1997) Nitrate leaching from cattle urine and faeces in Northeast USA, Soil Sci. Soc. Am. J. 61, 1787–1794.
- Stout W.L., Gburek W.J., Schnabel R.R., Folmar G.J., Weaver S.R. (1998) Soil-climate effects on nitrate leaching from cattle excreta, J. Environ. Qual. 27, 992–998.
- Stout W.L., Weaver S.R., Gbureck W.J., Folmar G.J., Schnabel R.R. (2000) Water quality implications of dairy slurry applied to cut pastures in northeast USA, Soil Use Manage. 16, 189–193.
- Thrash I. (1997) Infiltration rate of soil around drinking troughs in the Kruger National Park, South Africa, J. Arid Environ. 35, 617–625.
- Thrash I., Nel P.J., Theron G.K., Bothma J. du P. (1991) The impact of water provision for game on the herbaceous vegetation basal cover around a dam in Kruger National Park, Koedoe. 34, 121–130.
- Thurow T.L. (1991) Hydrology and erosion, in: Heitschmidt R.K., Stuth J.W. (Eds.), Grazing management: an ecological perspective, Timber Press, Portland, OR, pp. 141–160.
- Tisdale S.L., Nelson W.L. (1975) Soil fertility and fertilizers, Macmillan Publishing Co., Inc., New York, p. 694.

- USEPA (1988) National water quality inventory, United States Environmental Protection Agency Report to Congress, Washington, DC, Office of Water, US Govt, Print Office, p. 1990.
- USEPA (1979) Methods for chemical analysis of water and wastes, EPA-600/4-79-020, USEPA, Cincinnati.
- Van Horn H.H., Newton G.L., Kunkle K.E. (1996) Ruminant nutrition from an environmental perspective: factors affecting whole-farm nutrient balance, J. Anim. Sci. 74, 3082–3102.
- Warren S.D., Thurow T.L., Blackburn W.H., Garza N.E. (1986) The influence of livestock trampling under intensive rotation grazing and soil hydrologic characteristics, J. Range Manage. 39, 491–495.
- Wedin D.A., Tilman D. (1990) Species effects on nitrogen cycling: a test with perennial grasses, Oecologia 84, 433–441.
- White S.L., Sheffield R.E., Washburn S.P., King L.D., Green J.T. Jr. (2001) Spatial and time distribution of dairy cattle excreta in an intensive pasture systems, J. Environ. Qual. 30, 2180–2187.
- Yan T., Frost J.P., Keady T.W.J., Agnew R.E., Mayne C.S. (2007) Prediction of nitrogen excretion and urine of beef cattle offered diets containing grass silage, J. Anim. Sci. 85, 1982–1989.

# Biogeography of Soil Microbial Communities: A Review and a Description of the Ongoing French National Initiative

Lionel Ranjard, Samuel Dequiedt, Claudy Jolivet, Nicolas P.A. Saby, Jean Thioulouse, Jérome Harmand, Patrice Loisel, Alain Rapaport, Saliou Fall, Pascal Simonet, Richard Joffre, Nicolas Chemidlin-Prévost Bouré, Pierre-Alain Maron, Christophe Mougel, Manuel P. Martin, Benoît Toutain, Dominique Arrouays, and Philippe Lemanceau

### Contents

1	Intro	duction	858
2	Scien	tific Outcomes of Microbial Biogeography	858
	2.1	Phylogeography of Particular Populations	859
	2.2	Biogeography of Microbial Communities	859
	2.3	The First Concept of Microbial Biogeography:	
		"Beijerinck, 1913"	860
3	Soil I	Biodiversity Monitoring in Europe	860
	3.1	International Context	860
	3.2	The French Soil Survey: RMQS	861
	3.3	An Ongoing Project: The ECOMIC-RMQS	
		Project	861
	3.4	Need for Technical and Logistic Supports	863
4	Conc	lusion	864
Re	ferenc	ees	864

Abstract Microbial biogeography is the study of the distribution of microbial diversity on large scales of space and time. This science aims at understanding biodiversity regulation and its link with ecosystem biological functioning, goods and services such as maintenance of productivity, of soil and atmospheric quality, and of soil health. Although the initial concept dates from the early 20th century (Beijerinck (1913) De infusies en de ontdekking der backterien, in: Jaarboek van de Knoniklijke Akademie van Wetenschappen, Muller, Amsterdam), only recently an increasing number of studies have investigated the biogeographical patterns of soil microbial diversity. A such delay is due to the constraints of the microbial models, the need to develop relevant molecular and bioinformatic tools to assess microbial diversity, and the non-availability of an adequate sampling strategy. Consequently, the conclusions from microbial ecology studies have rarely been generally applicable and even the fundamental power-laws differ because the taxa-area relationship and the influence of global and distal parameters on the spatial distribution of microbial communities have not been examined. In this article we define and discuss the scientific, technical and operational limits and outcomes resulting from soil microbial biogeography together with the technical and logistical feasibility. The main results are that microbial communities are not stochastically distributed on a wide scale and that biogeographical patterns are more influenced by local parameters such as soil type and land use than by distal ones, e.g. climate and geomorphology, contrary to plants and animals. We then present the European soil biological survey network, focusing on the French national initiative and the "ECOMIC-RMQS" project. The objective

L. Ranjard (🖂)

INRA-Université de Bourgogne, UMR Microbiologie du Sol et de l'Environnement, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France

and

Platform GenoSol, INRA-Université de Bourgogne, CMSE, 17 rue Sully, B.V. 86510, 21065 Dijon Cedex, France e-mail: ranjard@dijon.inra.fr

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_37, © Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Ranjard et al., Agron. Sustain. Dev. 30 (2010) 359–365. DOI: 10.1051/agro:2009033

of the ECOMIC-RMQS project is to characterise the density and diversity of bacterial communities in all soils in the RMQS library in order to assess, for the first time, not only microbial biogeography across the whole of France but also the impact of land use on soil biodiversity (Réseau de Mesures de la Qualité des Sols = French Soil Quality Monitoring Network, 2200 soils covering all the French territory with a systematic grid of sampling). The scientific, technical and logistical outputs are examined with a view to the future prospects needed to develop this scientific domain and its applications in sustainable land use.

**Keywords** Soil biogeography • Microbial communities • Soil survey • Microbial ecology • Diversity

### 1 Introduction

Although microorganisms are the most diverse and abundant type of organism on earth (Gans et al., 2005; Curtis and Sloan, 2005), the determinism of microbial diversification and the distribution of microbial diversity from small to large scales has been poorly documented and is little understood. Most studies of prokaryote diversification have focused on variations due to mutations and/or lateral gene transfers and subsequent selection resulting from environmental stresses and competition for resources. Few have considered more neutral mechanisms, such as genetic drift due to physical isolation, thus revealing the crucial lack of integration of the spatial scale into microbial community assembly (Ranjard and Richaume, 2001; Papke and Ward, 2004).

Ecologists have long recognised that beta-diversity (how community composition changes across a landscape) is pivotal to understanding how environmental factors affect the magnitude and variability of biodiversity. This conceptual vision is also relevant to microorganisms since beta-diversity patterns offer valuable insights into the relative influence of dispersal limitations, environmental heterogeneity, and environmental and evolutionary changes in shaping the structure of ecological communities (Green et al., 2004). Although the spatial patterning of microbial diversity is known to have important consequences on plant community structure and ecosystem functioning, microbial betadiversity patterns have been poorly investigated and remain largely unknown. The empirical relationship between the number of species and the area sampled (taxa-area relationship) has not been empirically examined in microorganisms, as it has in plants and animals (Horner Devine et al., 2004; Green and Bohannan 2006). This may partly be explained by the characteristics of microorganisms, namely (i) their small size, which makes access within environmental matrices difficult, (ii) their high density (e.g. more than one billion per gram of soil) and (iii) their huge diversity (from 1000 to 1000000 species per gram of soil, Torsvik and Øvreås, 2002), not to mention the complexity of precisely defining their species. Progress is also hampered by difficulties in designing an adequate sampling strategy. Such a strategy needs to integrate a large scale of sampling (region, territory, etc.), with precise squaring that is representative of any landscape modifications, which therefore implies a very large number (several thousands) of samples.

Our aim in this paper is to present the results of the first analytical studies of the biogeography of soil microbial communities, the concept applied and the technical feasibility of this novel scientific domain in environmental microbiology. We shall then describe the European strategy and the French national soil survey in which different research teams with scientific expertise in soil science, statistics, microbial ecology and geochemistry are, for the first time, working to assess the inventory and mapping of soil microbial diversity on a national territorial scale.

## 2 Scientific Outcomes of Microbial Biogeography

Biogeography is the study of the distribution of biodiversity over space and time (Martiny et al., 2006). In other words, the aim in biogeography is to reveal where organisms live and their abundance, and to determine those environmental factors that select or maintain the presence of these organisms. This scientific approach was first applied during the eighteenth century in studies of the geographic distribution of plant and animal diversity. These investigations provided insights into the mechanisms that generate and maintain diversity in macroorganisms e.g., speciation, extinction, dispersal and species interactions (Brown and Lomolino, 1998).

Despite the key role of microorganisms in a wide range of biogeochemical cycles, few studies (in comparison with macroorganisms) have examined the distribution of microbial diversity on a broader scale than field plots. The first study describing and investigating microbial biogeography was conducted by Beijerinck (1913), who stated that "everything is everywhere, but, the environment selects" (see Sect. 3). Since then, few authors have examined the full extent of microbial diversity or described the biogeographical patterns in an attempt to assess this statement and specify which environmental factors exert the strongest influence on indigenous microbial communities. Even though recent advances in molecular biology have led to the development of tools to assess bacterial diversity in environmental samples without culturing, most studies have focused on cataloguing the bacterial diversity in particular sites or describing how bacterial communities have been affected by environmental perturbations (for review see Ranjard et al., 2000). Thus, the conclusions of such microbial ecology investigations cannot be generally applied as the data from different studies are difficult to compare and the trends deduced are often inconsistent.

# 2.1 Phylogeography of Particular Populations

To date, most studies dealing with microbial biogeography have been limited to the phylo-geography of particular populations, particularly of pathogenic or symbiotic organisms. One of the main results is that many host-associated microorganisms exhibit genetic and functional patterns that are related to the distribution of their hosts (for review see Martiny et al., 2006). As regards the free-living microorganisms, most recent investigations have been focused on individual soil bacterial strains (Cho and Tiedje, 2000). These studies have tended to demonstrate that the genetic distance between microorganisms is related to geographic distance, and have highlighted correlations between assembly composition and environmental or geographic characteristics (for review see Martiny et al., 2006). Few publications have considered the soil microbial community as a whole and how it is structured on a large spatial scale.

### 2.2 Biogeography of Microbial Communities

In one of the rare studies available, Green et al. (2004) genotyped fungal community structure in numerous Australian soils (about 1500) and were able to demonstrate that despite the high local diversity of microorganisms, the regional diversity was only moderate. Fierer and Jackson (2006) produced a continent-scale description of soil bacterial diversity by considering about 100 different soils sampled from the north to south of America. By applying a DNA fingerprinting method, they demonstrated that bacterial diversity was unrelated to site temperature, latitude and other variables that, in contrast, strongly influence plant and animal diversity, and that community composition was largely independent of geographic distance. The environmental factor most influencing bacterial diversity was soil pH, with the highest diversity occurring in neutral soils and the lowest in acidic soils. These studies also demonstrated that taxa-area relationships in soil microorganisms were weak, thus indicating that microbial biogeography differs fundamentally from that of "macroorganisms". By applying a pyrosequencing technique to ribosomal sequences in the same set of soil samples, Jones et al. (2009) defined the ecological attributes of particular populations such as Acidobacteria and confirmed the importance of soil pH in their dissemination.

In contrast, Johnson et al. (2003) demonstrated that variations in the genetic structure of bacterial communities from numerous agricultural soils were not correlated with pH but with soil texture and electrical conductivity. The overall inconsistency of these reports may possibly be ascribed to the inadequate sampling strategy in terms of number and representativeness of the soils sampled. However, it underlines the fact that the number of studies dealing with microbial-biogeography needs to be increased to understand the determinism of microbial diversity better, especially as this latter directly affects a wide range of ecosystem processes and therefore the quality of our environment.

# 2.3 The First Concept of Microbial Biogeography: "Beijerinck, 1913"

Microbial ecologists describing biodiversity on a large spatial scale, i.e. microbial biogeography, generally invoke one of the oldest fundamental paradigms in microbial ecology, "everything is everywhere, but, the environment selects", proposed by Beijerinck (1913). This tenet was used as the starting point for studies of prokaryotic biodiversity and their biogeographical patterns but was frequently misinterpreted. Baas Becking (1934), and more recently de Wit and Bouvier (2006), rehabilitated the original meaning of this statement, which reflects an apparent contradiction between empirical observations that specific microorganisms are observed in their characteristic environments and the idea that all microorganisms are cosmopolitan. Despite the technical difficulty of verifying the first statement, due to the detection limits of the approaches used to characterise microbial diversity, this premise implies that the genetic cohesiveness of prokaryotic populations can never be broken by physical isolation but solely by adaptation. These basic concepts were only implicit in the publication of Beijerinck (1913) but were tacitly accepted (Baas Becking, 1934).

The concept "everything is everywhere" is supported by several particularities of the microbial model: microorganisms (i) are small and easily transported, (ii) are able to form a resistant physiological stage that allows them to survive in hostile environments, and (iii) have extremely large population sizes with a high probability of dispersal and a low probability of local extinction (Fenchel, 2003). The fact that more than  $10^{18}$ – $10^{20}$  microorganisms are estimated to be transported annually through the atmosphere between continents supports the hypothesis of a wide dispersion of microbes (Gans et al., 2005). Further evidence is that bacteria can be isolated from places where "they should not be", e.g. thermophilic bacteria from cold sea water (Isaksen et al., 1994).

In contrast, "the environment selects" might seem to challenge the concept that "everything is everywhere", and suggests that geographic isolation of populations coupled with limited dispersal leads to allopatric speciation. An increasing number of studies have demonstrated that the physical isolation of freeliving microorganisms may be more widespread than previously thought (for review see Papke and Ward, 2004). However, most recent studies have been limited to characterisation of the culturable populations which are known to represent only a very small fraction of the whole community (Amann et al., 1995). To date, few studies have focused on microbial communities, possibly because of the technical limitations associated with identifying the huge microbial diversity in natural ecosystems and the difficulties in detecting minor populations. These technological limitations have now been partly resolved thanks to the recent development of molecular tools that circumvent the isolation and culture of organisms, and allow microbial community structure and diversity to be characterised without a priori knowledge (for review see Ranjard et al., 2000; Christen, 2008). Furthermore, these tools are now generally automated and allow the moderate throughput essential to studies involving the characterisation of numerous environmental samples.

### **3** Soil Biodiversity Monitoring in Europe

### 3.1 International Context

Although soil inventory programmes exist in all European countries, there are not many fully operational soil monitoring systems in Europe (Morvan 2008). Few include more than one sampling point in time so that most are mere inventories. The only EUwide soil monitoring network is the ICP forest level 1 grid, which was partly re-sampled in 2006-2007 within the Forest Focus BioSoil project (Lacarce et al., 2009). The data requirements for this re-sampling did not include microbial diversity. Progress in recent years has been hampered by a lack of perception of the importance of soil, data ownership issues and data incompatibility resulting from the multiplicity of different sampling and analytical procedures. The EN-VASSO project addressed the need to characterise soils by setting up a series of interlinked objectives to harmonise the soil data sets that currently exist in EU Member States (Kibblewhite et al., 2008). Eight threats to soil (erosion, declining organic matter, contamination, compaction, salinisation, loss of biodiversity, soil sealing, landslides and flooding) have been identified in the European Commission's official Soil Communication part of the Thematic Strategy for Soil Protection in Europe. The aim of the ENVASSO project was to develop a system to harmonise existing, mostly

national, data sets and provide a central reference point to assess current soil status and ensure sustainable management in the future. This project identified the physical and chemical parameters to monitor, most of which were already being monitored by the national soil surveys conducted in different European countries.

### 3.2 The French Soil Survey: RMQS

It was apparent from the initial data collected during the ENVASSO project that biodiversity was not fully included in any national soil monitoring network except in the Netherlands. In 2001, a new structure was created in France to reorganise soil mapping and soil monitoring programmes and to provide relevant insights into the spatial distribution of soils and the evolution of their properties. This structure, called the Scientific Group on Soils (GIS Sol) includes the Ministries of Agriculture and Environment, the Environment and Energy Management Agency (ADEME), the Research Institute for Development (IRD), the National Forest Inventory (IFN) and the National Institute of Agronomic Research (INRA). The main soil monitoring programme is the French Soil Quality Network ("Réseau de Mesures de la Qualité des Sols", RMQS), that is based on a 16 km by 16 km grid, with 2200 sites representative of the main soil systems and land uses in France. This configuration has the advantage of being fully compatible with the sites of the ICP forest level 1 network (now called BioSoil) which monitors forested soils across Europe. The RMQS is a complete and balanced network of 2200 sites which will be sampled every 10 years to monitor soil quality. The primary objectives of the network are to characterise and quantify diffuse contamination in trace elements (Saby et al., 2006) and to evaluate and monitor organic carbon stocks. For this purpose, a set of analyses (particle-size distribution, bulk density, C, N, pH, trace elements, etc.) and a complete description of the soil profile at each site are obtained together with information about past activities, the environment, etc. Apart from these objectives, the network is responsible for many other soil quality evaluations and monitoring, e.g. persistent organic pollutants (pesticides, dioxins, organochlorides, PAH, etc.).

## 3.3 An Ongoing Project: The ECOMIC-RMQS Project

Based on this soil survey, the scientific project "ECOMIC-RMQS" was set up in 2006, offering the first opportunity to implement biological diversity in a soil monitoring network. This project is coordinated by the Microbiology of the Soil and Environment Centre ("Centre de Microbiologie du Sol et de l'Environnement" i.e. CMSE, INRA Dijon, Burgundy, France) and is one of the first steps to demonstrate the technical feasibility and scientific relevance of federating European initiatives to monitor soil biodiversity monitoring.

The objective of the ECOMIC-RMQS project is to characterise the density, genetic structure and diversity of bacterial communities in all soils in the RMQS library (2200 soils sampled up to 2009) in order to assess, for the first time, not only microbial biogeography across the whole of France but also the impact of land use on soil biodiversity. The strategy for characterising the density and diversity of the bacterial communities relies on molecular tools such as quantitative PCR, DNA micro-array and DNA fingerprinting on soil DNA extracts (Fig. 1).

This integrated project will provide cognitive insights into the ecological theory on the community assembly by:

- elucidating the relative contribution of geographic isolation versus wide dispersal in limiting bacterial diversification,
- allowing better examination of the taxa-area relationship for bacteria,
- deciphering the hierarchy of the environmental parameters (plant cover, physico-chemical characteristics, climate factors, etc.) that most contribute to bacterial community diversity and composition.

This project should also have more applied outcomes as a result of:

- determining the state of bacterial diversity in French soils,
- better estimation of the impact of land use and human activities on microbial diversity and distribution,
- identifying bacterial bio-indicators specific to land management and human activities.

**RMQS :** « Réseau de Mesure de la Qualité des Sols » = « French Soil Quality Monitoring Network » managed by unit INFOSOL, (INRA Orléans), soil systematic sampling grid: 16 kmx16km square, covering all the French territory  $\Rightarrow$  2,200 soils



Fig. 1 Schematic description of strategy and outputs of the ECOMIC-RMQS project

**Density/Species number** 

The ECOMIC-RMQS project provides an appropriate means of (i) assessing the microbial-biogeography by use of the RMQS and (ii) elucidating the determinisms of bacterial community diversification in soils better. This project should provide answers to some of the questions originating from mechanistic hypotheses:

- Are microbial communities a "black box" with no spatial structure or, like macroorganisms, do they exhibit a particular distribution with predictable, aggregated patterns from local to regional scales? In other words, does a taxa-area relationship exist in microbial-biogeography? (Green et al., 2004; Horner Devine et al., 2004).
- Are spatial variations due to contemporary environmental factors or to historical land use and contingencies?
- Which environmental factors (edaphic, climatic, land use, anthropogenic) most contribute to the structure and diversity of bacterial communities in soil on very broad geographic scales?

The first results obtained in this programme have demonstrated:

- a non-stochastic distribution of bacterial diversity which is spatially structured in biogeographical patterns on a regional scale (Dequiedt et al., 2009),
- a taxa-area relationship for soil bacterial communities,
- a positive correlation between bacterial diversification and landscape diversity and fragmentation,
- a greater influence of local environmental local parameters (pedoclimatic and land use) as opposed to global parameters (climate and geomorphology) on bacterial community density and diversity,
- the strongly deleterious effects that specific land use (especially agriculture) can have on indigenous bacterial communities.

# 3.4 Need for Technical and Logistic Supports

One of the main prerequisites of a project such as ECOMIC-RMQS, which is aimed to assess microbial diversity on a wide scale, is the development of suitable logistics for storing and managing very large numbers of biological samples and associated data. In addition, the soil samples need to be characterised under medium-throughput conditions with high levels of reproducibility and quality of the applied procedures. The GenoSol Platform, which was set up to store and manage soil samples as well as to ensure molecular characterisation of the microbial communities, meets these requirements (Ranjard et al., 2009). Thus, two complementary structures described as Soil Biological Resource Centres have been developed in France: the INFOSOL Conservatory and GenoSol Platform.

The INFOSOL unit in Orléans (France) houses the soil conservatory. This soil library has been designed for the management, storage, preparation and dissemination of soil samples from the RMQS programme in the long term. Thousands of large samples (5 to 10 kg each) are stored air-dried, under controlled conditions of temperature and hygrometry. The aim is to provide (i) a memory of the state of French soil, (ii) reference samples for re-analysis after other campaigns, and (iii) a soil bank for research development.

The GenoSol Platform (http://www.dijon.inra.fr/ plateforme\_genosol) was created in 2008 by the "Centre de Microbiologie du Sol et de l'Environnement" (CMSE, INRA Dijon, Burgundy, France). The aim is to provide an appropriate logistic structure for the acquisition, storage and characterisation of soil genetic resources obtained by extensive sampling (several hundred to several thousand soils), on very large space and/or time scales (national soil survey, long-term experimental sites), and to make these resources readily available to the scientific community and to policy-makers. The ultimate goal is to produce a reliable reference system based on molecular characterisation (taxonomic and functional features) of soil microbial communities, that facilitates the scientific interpretation of sample analyses on large scales of time and space sampling. Another aim of the platform is the long-term storage of a library of soil genetic resources (soil DNA) to be made available to national and international scientific communities (Ranjard et al., 2009). In summary, the GenoSol Platform can be considered as a logistic and technical tool and therefore as a strategic partner for research units who wish to benefit from large-scale soil sampling without needing to develop cumbersome or circumstantial methodologies or organisations.

In the context of the ECOMIC-RMQS project, the GenoSol Platform has:

- built up and maintains a national soil DNA library which is to be made available to the scientific community as a whole in order to assess microbial diversity in the future with more powerful tools and/or other molecular analysis,
- developed in collaboration with the INFOSOL conservatory a database of the genetic structure and taxonomic diversity of microbial communities in French soils, so that an Atlas of Soil Biodiversity can be compiled on a national scale,
- established a reference frame for interpreting these analyses by (i) treating the determinants of variability of the microbial communities in soils on a hierarchical basis, and (ii) quantifying the impact of human activities on these communities.

### **4** Conclusion

Various lines in the studies of microbial biogeography currently available could be further developed:

- the scale of investigation could be increased by considering a whole territory or continent to permit better comparison of the different pedoclimatic regions,
- widescale soil sampling could be conducted in the long term to evaluate the influence of land use management and global changes on the evolution of soil biodiversity,
- the characterisation of soil microbial diversity could be improved by applying recent and innovative techniques such as pyrosequencing of ribosomal genes to permit exhaustive evaluation of the taxonomic diversity of indigenous communities (Christen, 2008; Roesch et al., 2007),
- soil biodiversity could be linked with soil functioning to improve management and protection of the various resources and services of soil,
- the range of variations in diversity could be described for a given pedoclimatic zone and within this zone, for a given land use. Such information could be used to interpret soil quality analyses for soil users (farmer, industrial or urban), biodiversity erosion, and to improve land use management with a view to sustainable development.

Microbial biogeography is therefore relevant at a landscape level to understanding soil biodiversity erosion brought about by natural and/or anthropogenic activities better. Even if the relationship between soil biodiversity and soil or ecosystem functions remains incomplete, such a research strategy should improve soil management in the current context of increasing ecosystem goods and services. Consequently, at the European level, a better coordination of the different national soil surveys and of the strategy employed to characterise biodiversity constitutes a challenge for the future soil protection policy in terms of sustainable development.

Acknowledgements The ECOMIC-RMQS programme was supported by the French National Research Agency (ANR) and ADEME. Sampling was supported by a French Scientific Group of Interest on soils: the "GIS Sol", involving the French Ministry for Ecology and Sustainable Development (MEDD), the French Ministry of Agriculture (MAP), the French Institute for the Environment (IFEN), the French Environment and Energy Management Agency (ADEME), and the National Institute for Agronomic Research (INRA). We thank all the soil surveyors and technical assistants involved in sampling the sites.

### References

- Amann R.I., Ludwig W., Scheider K.H. (1995) Phylogenetic identification and in situ detection of individual microbial cells without cultivation, Microbiol. Rev. 59, 143–169.
- Bass Becking L.G.M. (1934) Geobiologie of inleiding tot de milieukunde, Van Stockum & Zoon, The Hague.
- Beijerinck M.W. (1913) De infusies en de ontdekking der backterien, in: Jaarboek van de Knoniklijke Akademie van Wetenschappen, Muller, Amsterdam.
- Brown J.H., Lomolino M.V. (1998) Biogeography, Sinauer, Sunderland.
- Cho J.C., Tiedje J. (2000) Biogeography and degree of endemicity of Fluorescent *Pseudomonas* strains in soil, Appl. Environ. Microbiol. 66, 5448–5456.
- Christen R. (2008) Global sequencing: a review of current molecular data and new methods available to assess microbial diversity, Microb. Environ. 23, 253–268.
- Curtis T.P., Sloan W.T. (2005) Exploring microbial diversity A vast below, Science 309, 1331–1333.
- de Wit R., Bouvier T. (2006) 'Everything is everywhere, but, the environment selects'; what did Baas Becking and Beijerinck really say? Environ. Microbiol. 8, 755–758.
- Dequiedt S., Thioulouse J., Jolivet C., Saby N.P.A., Lelievre M., Maron P.A., Martin M.P., Chemidlin-Prévost-Bouré N., Toutain B., Arrouays D., Lemanceau P., Ranjard L. (2009) Biogeographical patterns of soil bacterial communities, Environ. Microbiol. Report 1, 251–255.

- Fenchel T. (2003) Biogeography for bacteria, Science 301, 925–926.
- Fierer N., Jackson R.B. (2006) The diversity and biogeography of soil bacterial communities, Proc. Natl Acad. Sci. (USA) 103, 626–631.
- Gans J., Wolinsky M., Dunbar J. (2005) Computational improvements reveal great bacterial diversity and high metal toxicity in soil, Science 309, 1387–1390.
- Green J.L., Bohannan B.J.M. (2006) Spatial scaling of microbial diversity, Trends Ecol. Evol. 21, 501–507.
- Green J.L., Holmes A.J., Westoby M., Oliver I., Briscoe D., Dangerfield M., Gillings M., Beatlle A.J. (2004) Spatial scaling of microbial eukaryote diversity, Nature 432, 747–750.
- Horner-Devine M.C., Lage M., Hughes J.B., Bohannan B.J.M. (2004) A taxa-area relationship for bacteria, Nature 432, 750–753.
- Isaksen M.F., Bak F., Jorgensen B.B. (1994) Thermophilic sulfate reducing bacteria in cold marine sediment, FEMS Microbiol. Ecol. 14, 1–8.
- Johnson M.J., Lee K.Y., Scow K.M. (2003) DNA fingerprinting reveals links among agricultural crops, soil properties, and the composition of soil microbial communities, Geoderma 114, 279–303.
- Jones R.T., Robeson M.S., Lauber C.L., Hamady M., Knight R., Fierer N. (2009) A comprehensive survey of soil acidobacterial diversity using pyrosequencing and clone library analyses, ISME J. 17, 1–12.
- Kibblewhite M.G., Jones R.J.A, Montanarella L., Baritz R., Huber S., Arrouays D., Micheli E., Stephens M. (2008) Environmental Assessment of Soil for Monitoring: Volume VI. Soil Monitoring System for Europe, EUR 23490 EN/6, Office for the Official Publications of the European Communities, Luxembourg, 188 p.
- Lacarce E., Le Bas C., Cousin J.-L., Pesty B., Toutain B., Houston Durrant T., Montanarella L. (2009) Data

management for monitoring forest soils in Europe for the Biosoil project, Soil Use Manage. 25, 57–65.

- Martiny J.B.H., Bohannan B.J.M., Brown J.H., Colwell R.K., Furhman J.A., Green J.L., Horner-devine M.C., Kane M., Krumins J.A., Kuske C.R., Morin P.J., Naeem S., Ovreas L., Reysenbach A.L., Smith V.H., Staley J.T. (2006) Microbial biogeography: putting microorganisms on the map, Nature 4, 102–112.
- Morvan X.P.P., Saby N.P.A., Arrouays D., Le Bas C., Jones R.J.A., Verheijen F.G.A., Bellamy P.H., Stephens M., Kibblewhite M.G. (2008) Soil monitoring in Europe: a review of existing systems and requirements for harmonisation, Sci. Total Environ. 391, 1–12.
- Papke R.T., Ward D.M. (2004) The importance of physical isolation to microbial diversification, FEMS Microbiol. Ecol. 48, 293–303.
- Ranjard L., Dequiedt S., Lelievre M., Maron P.A., Mougel C., Morin F., Lemanceau P. (2009) Platform GenoSol: a new tool for conserving and exploring soil microbial diversity, Environ. Microbiol. Report 1, 97–99.
- Ranjard L., Poly F., Nazaret S. (2000) Monitoring complex bacterial communities using culture-independent molecular techniques: application to soil environment, Res. Microbiol. 151, 167–177.
- Ranjard L., Richaume A. (2001) Quantitative and qualitative microscale distribution of bacteria in soil, Res. Microbiol. 152, 707–716.
- Roesch L., Fulthorpe R.R., Riva A., Casella G., Hadwin A., Kent A. (2007) Pyrosequencing enumerates and contrast soil biodiversity, ISME J. 1, 283–290.
- Saby N., Arrouays D., Jolivet C., Boulonne L., Pochot A. (2006) Geostatistical assessment of lead in soil around Paris, France, Sci. Total Environ. 367, 212–221.
- Torsvik V., Øvreås L. (2002) Microbial diversity and function in soil: from genes to ecosystems, Curr. Opin. Microbiol. 5, 240–245.

# Part VII Alternative Fertilisation
# **Nitrogen Rhizodeposition of Legumes**

Joëlle Fustec, Fabien Lesuffleur, Stéphanie Mahieu, and Jean-Bernard Cliquet

### Contents

1	Intro	duction	870
2	Biolog	gical N <sub>2</sub> Fixation of Legumes	870
	2.1	Processes	870
	2.2	Benefits and Use of Legumes in Agrosystems	871
3	Quan	tification of N Rhizodeposition	872
	3.1	Estimation of Below-Ground N	872
	3.2	<sup>15</sup> N Labelling Methods	873
	3.3	Amounts of N Rhizodeposited by Legumes	877
4	N Rhi	izodeposition Pathways	877
	4.1	General Considerations	877
	4.2	Senescence of Roots and Nodules	878
	4.3	Exudation of Soluble Compounds	878
5	Conc	usion	878
Re	ferenc	es	879

J. Fustec (🖂)

Abstract Because nitrogen is one of the major elements limiting growth of plants in agrosystems, large amounts of N fertilisers have been used in the second half of the twentieth century. Chemical fertilisers have contributed to increasing crop yields and food supply, but they have induced environmental damage such as nitrate pollution and wasting fossil fuel. The use of legumes grown in rotations or intercropping is now regarded as an alternative and sustainable way of introducing N into lower input agrosystems. Here we review agricultural practices, measurement methods and biological pathways involved in N cycling. We show that plant roots interact intimately with soil microflora to convert the most abundant but relatively inert form of N, atmospheric N<sub>2</sub>, into biological substrates available for growth of other plants, through two consecutive processes; namely, N2 fixation and N rhizodeposition. In intercropping, companion plants benefit from biological fixation by legumes and subsequent transfer of N from legumes to non-legumes. This transfer from legumes to the release of N compounds by legume roots, a process named rhizodeposition, then the uptake by the companion crop. The two main rhizodeposition pathways are (i) decomposition and decay of nodules and root cells, and (ii) exudation of soluble N compounds by plant roots. The contribution of root N and rhizodeposited N to the soil-N pool is difficult to measure, particularly in the field. Firstly, root N is often underestimated because root recovery is problematic. Second, assessment of N rhizodeposition is challenging. Several <sup>15</sup>N labelling methods have been performed for different legume species. Rhizodeposition of N, as a percentage of total plant N, varied from 4 to 71%. The high variability of the results illustrates the need for more studies of

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_38,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Fustec et al., Agron. Sustain. Dev. 30 (2010) 57–66. DOI: 10.1051/agro:2009003

LUNAM University, Groupe ESA, UR Laboratoire d'Ecophysiologie Végétale et Agroécologie, 55 rue Rabelais, 49007 Angers Cedex 01, France e-mail: j.fustec@groupe-esa.com

the environmental and genetic factors influencing the amount of N rhizodeposits released by legumes under field conditions.

**Keywords** N rhizodeposition  $\cdot$  Legumes  $\cdot$  N<sub>2</sub>fixation  $\cdot$ <sup>15</sup>N  $\cdot$  Isotopic methods  $\cdot$  Root exudates  $\cdot$  Ecological fertilisation

## 1 Introduction

Even though N is among the most abundant elements on earth, it is also the major element limiting growth of plants in many agricultural systems because of its unavailability for plants (Hartwig, 1998; Vance, 2001). N fertilisers have been considered for many years as a reasonable insurance against yield loss and have been used extensively (Vance, 2001) but contribute substantially to environmental pollution (Deutsch et al., 2006; Umar and Iqbal, 2007). It is now established that excessive use of these fertilisers affects the balance of the nitrogen cycle in soils, causes eutrophication because of nitrate leaching and has contributed to global warming because of gaseous loss as N2O. The non-stop use of N fertilisers would also accelerate the depletion of stocks of non-renewable energy resources required for fertiliser production. Furthermore, there are vast areas in the developing world where N fertilisers are neither available nor affordable due to weak infrastructure, poor transportation and high cost.

These problems explain why biological alternatives using diazotrophic prokariots have become of increasing interest in agricultural practices in the last few years, particularly for low-input systems. Biological N fixation can act as a sustainable source of N and can complement or replace fertiliser inputs (Garg and Geetanjali, 2007). The two main agricultural practices to benefit from biological N fixation, crop rotation and intercropping legumes (Fabaceae), and non-fixing plants, were practised in ancient times, even if the basis for the benefit derived was not understood (Burris, 1974). Most of the fixed N in legumes is harvested and fed to animals, but evidence from a number of experiments using different methodologies indicates that legumes can deposit significant amounts of N in the soil during growth (Jensen, 1996a, b; McNeill et al., 1998; Khan et al., 2002a; Mahieu et al.,

2007; Wichern et al., 2007a, b). Fixed N can also be transferred to intercropped non-legumes in the case of mixed cropping systems, or to following crops in the case of crop rotation.

In addition to the use of legumes in agriculture, other technologies to take advantage of  $N_2$ -fixing micro-organisms include the utilisation of the symbiosis between the fern *Azolla azollae* and the  $N_2$ -fixing cyanobacterium *Anabaena azollae* as a green manure in rice wetlands, and the use of free-living  $N_2$ -fixing bacteria such as *Azopirillum* inoculated into the rhizosphere of grasses.

## 2 Biological N<sub>2</sub> Fixation of Legumes

#### 2.1 Processes

Nitrogen fixation is carried out by a small number of diazotrophic prokaryotic microorganisms, belonging to a wide range of eubacteria and archaebacteria. Diazotrophs are usually divided into free-living and symbiotic forms, though some cyanobacteria are able to fix N either independently or in symbiotic association. Symbiotic diazotrophs include a number of genera of the Rhizobiaceae, which form the well-documented symbiosis with legumes (Gordon et al., 2001; Garg and Geetanjali, 2007), where nitrogen fixation takes place in specialised organs, the nodules. Most of these nodules are formed on legume roots but some rhizobia such as Azorhizobium caulinodans are able to form stem nodules. Symbiotic N2 fixation in legumes is the result of a structurally and physiologically highly organised, host-specific mutualistic interaction between rhizobia and legumes. Biological N fixation is catalysed by an anaerobic enzyme, nitrogenase, which carries a complex metallocluster on its active site. The most abundant nitrogenase contains iron and molybdenum at this site but others contain iron and vanadium, or iron only when molybdenum is not available (Curatti et al., 2006). Because of the economic and ecological benefits of N2 fixation, the genes associated with this process, designated nif, have been extensively studied and are now well characterised (Gordon et al., 2001). Because nitrogenase is inhibited upon exposure to oxygen, nitrogen-fixing organisms have certain adaptations. In the case of the legume-Rhizobium

symbiosis, the two main adaptations are the formation of the oxygen diffusion barrier into the nodule and the synthesis of the oxygen carrier protein in the symbiosome, leghaemoglobin (Gordon et al., 2001). A number of other non-legume plants, mainly woody species, also produce N<sub>2</sub>-fixing nodules, in symbiosis with the actinomycete, *Frankia* (Uselman et al., 1999).

# 2.2 Benefits and Use of Legumes in Agrosystems

Because of their ability to fix N<sub>2</sub>, legumes are considered to be involved in ecological facilitation processes in all ecosystems (Loreau and Hector, 2001; Rochon et al., 2004; Padilla and Pugnaire, 2006). A wide range of legumes are grown around the world, for production of protein-rich seeds or for harvest of the whole shoot. In agrosystems, legumes contribute nitrogen benefits in two main ways:

(i) Legumes are N-rich plants which can be used in crop rotations to increase the soil-N pool (Chalk, 1998). For this purpose, several legume species such as clovers (Trifolium sp.), alfafa and vetches (Vicia sativa L. and other Vicia genera), fenugreek (Trigonella foenum-graecum L.), lupin (Lupinus angustifolius L.), velvet bean (Mucunia pruriens Bak.), Crotalaria spectabilis Roth., or Sesbania rostrata Brem. are included in rotations and used as green manure. They contribute to nutrient cycling, soil organic matter conservation, and to the nutrient supply for succeeding crops. However, numerous legumes included in rotations are grain legumes. They are grown worldwide and Crépon (2006) reported production of  $241 \times 10^6$  t of dry matter in the 2003/2004 season. Soybean (Glycine max L.) is mainly produced in North and South America and in Asia. Pea (Pisum sativum L.), fababean (Vicia faba ssp minor L.) and dry bean (Phaseolus vulgaris L.) are mainly produced for feed in the northern hemisphere, since in the southern hemisphere, the most common grain legumes are mainly grown for food and are dry bean, chickpea (Cicer arietinum L.) and cowpea (Vigna unguiculata L.). Lentil (Lens esculenta L.), pigeon pea (Cajanus cajan L.) and peanut (Arachis hypogea L.) are also commonly used for human food. Nitrogen harvest indices of grain

legumes such as soybean, pea, fababean or lupin are often high; for instance, N accumulated in the seeds may represent more than 85% of plant N for soybean (Toomsan et al., 1995), and more than 75% for pea plants (Mahieu et al., 2007).

Since roots and rhizodeposits are so rich in N, including a grain legume in rotations may lead to a positive N-preceding effect on the following crop, despite N losses due to harvest. Hence, compared with a cereal grown in the same conditions, greater levels of inorganic N are recorded after harvesting grain legumes, especially in deeper soil layers (Crozat and Fustec, 2006). However, soil inorganic N measurements do not take into account changes in the organic N pool.

(ii) Legumes grown simultaneously and in the same field as non-fixing species (intercropping) lead to a more efficient use of soil resources in time and space (Loreau and Hector, 2001; Hauggaard-Nielsen and Jensen, 2005; Corre-Hellou et al., 2006). The niche separation effect often results in a higher yield in an intercrop than in a sole crop for the nonfixing species. In mixtures with non-fixing plants,  $N_2$  fixation by legumes is higher than in a monoculture regardless of management or location (Carlsson and Huss-Danell, 2003; Corre-Hellou et al., 2006). Experiments undertaken using mixtures of annual crops (for instance, pea-barley intercropping) have shown that this effect is higher in low-input systems than in others, and leads to more stable yields in problematic environments (Jensen, 2006; Corre-Hellou et al., 2007).

Both the niche complementarity effect (Loreau and Hector, 2001) and soil N-pool increase can benefit perennial cover such as legume-based grasslands (Soussana and Machado, 2000; Høgh-Jensen, 2006; Rasmussen et al., 2007). Forage legumes are widespread and have the potential to give high yields over a range of climatic conditions; the four major forage legumes alfafa (Medicago sativa L.), red clover (Trifolium pratense L.), subterraneum clover (T. subterraneum L.) and white clover (T. repens L.) together cover grassland of hot and dry regions of the earth (Frame et al., 1998). While white clover is the most widespread clover used in agriculture, birdsfoot trefoil (Lotus corniculatus L.) is also abundantly sown in temperate and northern areas, as is, to a lesser extent, alsike clover (T. hybridum L.).

## **3** Quantification of N Rhizodeposition

#### 3.1 Estimation of Below-Ground N

When legumes are used as green manure, biological fixation of N can be transferred to the soil through decomposition of above- and below-ground legume residues after harvest (Fujita et al., 1992). This is the reason why legumes are used in organic agriculture and are undersown with cereals for subsequent incorporation into the soil as green manure. Additionally, in intercropping systems, legume roots also release a significant proportion of N into the rhizosphere (rhizodeposited N). However, studies dealing with N balance in rotational farming systems including legume crops have long omitted to consider the below-ground contribution of legumes to the soil-N pool (Unkovich and Pate, 2000). The below-ground N pool can be defined as the sum of visible fibrous macro-root N and the part of soil N derived from rhizodeposition (Høgh-Jensen and Schjoerring, 2001). Estimation of soil N derived from rhizodeposition is greatly influenced by the method of measurement. Sampling of the roots and soil has major consequences on the results. Two kinds of methods are available for measuring below-ground N, with or without the use of a <sup>15</sup>N isotope:

(i) The most simple and commonly used approach for assessing below-ground N involves physical removal of the roots from the soil. Using this method, values of below-ground N as a percentage of total plant N are often very low compared with those obtained in the greenhouse. This is probably because sampling the entire root biomass is challenging, as many roots are fine and fragile and difficult to recover by wet sieving (Bergersen et al., 1989; Chapman and Myers, 1987; Toomsan et al., 1995; Russell and Fillery, 1996b; Dalal et al., 1997; Rochester et al., 1998; Unkovich and Pate, 2000). Greenhouse experiments undertaken in pots with limited volume allow a higher recovery of the root compartment (Mahieu et al., 2007). In addition, physical recovery of roots does not take rhizodeposited N into account, though this is also a necessary value for assessing below-ground N (Khan et al., 2002a, b).

Crawford et al. (1997) used a sequential coring and summation technique proposed by Hansson and Steen (1984) designed to assess total root production from repeated and simultaneous measurements of living roots, dead roots and old organic material. This method seems more accurate than assessments based solely on physical recovery of intact roots, but total root biomass remains underestimated.

(ii) Direct labelling of legumes with a tracer such as  $^{15}N$  provides a means to assess the two components of below-ground N and particularly rhizodeposited N in the soil.  $^{15}N$  is applied to a part of the plant and transferred to all organs by the sap stream, so rhizodeposits are  $^{15}N$ -enriched (Figs. 1 and 2). The percentage of NdfR (N derived from rhizodeposition) is usually calculated using equation (1), proposed by Janzen and Bruinsma (1989):

%NdfR = [(atom% <sup>15</sup>Nexcess soil)/

 $(atom\%^{15}Nexcess root)] \times 100$  (1)

The atom% <sup>15</sup>N excess values were obtained by correcting the <sup>15</sup>N enrichments with background values.

%NdfR=[(atom%<sup>15</sup>Nsoil enriched-soil background)/

 $(atom\%^{15}Nroot enriched - root background)] \times 100$ (2)

The <sup>15</sup>N abundance of plants grown in unlabelled soil, or of unlabelled legume or non-legume control plants, has often been used as background (Jensen, 1996a; Russell and Fillery, 1996a, b; Khan et al., 2002a, b; Mayer et al., 2003; Mahieu et al., 2007; Gylfadóttir, 2007). Schmidtke (2005a) has demonstrated that the lower the <sup>15</sup>N abundance of the roots, the more important the choice of adequate soil and root background values (Eq. (2)). The best estimation of N derived from rhizodeposition is obtained when the <sup>15</sup>N abundance of soil unlabelled N is used for soil background and the <sup>15</sup>N abundance of unlabelled roots for root background. Non-fixing plants can also be used for soil and root background values. As N re-absorptions are not taken into account, equations (1, 2) correspond to the assessment of net N rhizodeposition. The amount of total N (mg) derived from rhizodeposition is calculated by multiplying the N amount in this pool with the % NdfR value.

Root/soil sampling methods may also influence the results, since they have direct consequences on <sup>15</sup>N enrichment values of roots and soil. In some studies, roots are separated from the soil by dry gentle sieving (2 mm) before being carefully brushed to give a clean root fraction (McNeill et al., 1997, 1998; McNeill and Fillery, 2008). After root/soil sieving, Yasmin et al.



**Fig. 1** Some techniques used to assess below-ground contribution of legumes to the soil-N pool in the field. (1) – In the <sup>15</sup>N dilution technique, <sup>15</sup>N is provided directly to the soil, but the differences in <sup>15</sup>N natural abundance between air and soil can also be used. (2) – <sup>15</sup>N<sub>2</sub> enrichment technique: nodulated roots are exposed to <sup>15</sup>N<sub>2</sub>. (3) and (4) – In the leaf-feeding techniques and

in the cotton-wick technique, <sup>15</sup>N is provided as urea, nitrate or ammonium contained in a vial, to the above-ground parts. With the cotton-wick technique, <sup>15</sup>N-labelling solution is provided to the plant by means of a cotton-wick passing through a hole in the plant stem. (See Fig. 2 for split-root technique)

(2006) separated the roots from the remaining adhering soil (called 'rhizosphere soil') by -40°C freezedrying for 2 d. In other studies (Sawatsky and Soper, 1991; Mayer et al., 2003; Schmidtke, 2005a, b; Mahieu et al., 2007; Wichern et al., 2008), after root collection by gentle dry sieving, visible micro-roots were handcollected with tweezers. Then all roots were shaken in a closed dish with deionised water, and the rinse solution was pooled with the soil sample. Most experiments are undertaken in a sandy substrate to make soil/root sorting easier. In the field or under rain shelters, plants are often planted in columns pushed down into the soil (or mesotrons; Russell and Fillery, 1996b; McNeill et al., 1997; Gylfadóttir, 2007; Table 1), or in microplots delimited with plastic or iron sheets (Rochester et al., 1998; Mahieu et al., 2007).

The use of equation (1) assumes a uniform distribution of <sup>15</sup>N label throughout the root system and similar enrichments of both N deposited and of roots, but differences in <sup>15</sup>N enrichment between fine roots, coarse roots and nodules are often observed (Khan et al., 2002a,b; Russell and Fillery, 1996a; McNeill and Fillery, 2008).

## 3.2 <sup>15</sup>N Labelling Methods

Isotopic methods should ideally allow a uniform labelling of the whole plant. The <sup>15</sup>N label used to assess below-ground N can be provided to the legume in different ways.

(i) In *the* <sup>15</sup>*N dilution technique*, the label is provided directly to the soil and N fixation is estimated by the input of <sup>14</sup>N from the atmosphere (Fig. 1). This method is reliable for measurement of N<sub>2</sub> fixation by legumes and transfer to companion plants (Giller et al., 1991; Moyer-Henry et al., 2006; Paynel et al., 2008) but is strongly influenced by small differences in the spatial and temporal distribution of soil <sup>15</sup>N when used for measurement of N rhizodeposition (Hétier et al., 1986; Khan et al., 2007). Poth et al. (1986) used a soil with very low nitrogen content and labelled this soil with <sup>15</sup>NH<sub>4</sub> for six years to increase the accuracy of the measurement of rhizodeposition by pigeonpea plants in a greenhouse study.

(ii) The  ${}^{15}N_2$  enrichment technique, by which nodulated roots are exposed to  ${}^{15}N_2$ , is the more direct way to measure the input of fixed N<sub>2</sub> into the rhizosphere



**Fig. 2** Split-root method applied to a pea plant. Experimental design used by Mahieu et al. (2007). (1) Hydroponic labelling compartment filled with clay marbles and nutrient solution containing  ${}^{15}NO_3 - {}^{15}NH_4$ . (2) Compartment filled with soil in which net N rhizodeposition was measured. (3) Rubber tubes protecting the upper part of the roots from desiccation. (4) Putty

(Fig. 1; Warembourg et al., 1982; McNeill et al., 1994; Russelle et al., 1994). However, this technique requires specific equipment and cannot be applied easily in the field. Furthermore, free-living  $N_2$ -fixing bacteria can use <sup>15</sup>N and complicate interpretation of results.

(iii) *The leaf-feeding technique* involves feeding  $^{15}N$  as a gas (NH<sub>3</sub>), or as a solution (urea, ammonium or nitrate; Fig. 1). Janzen and Bruinsma (1989) exposed shoots of wheat plants (*Triticum aestivum* L.) to  $^{15}NH_3$  for a relatively short duration (6 h) periodically during the growing season. For this purpose, plants were temporarily placed in a sealed

plexiglass enclosure, and the medium was sealed from the atmosphere. This method resulted in a uniform labelling of the above- and below-ground parts (though <sup>15</sup>N enrichment in the roots was lower than in the shoots), but has not been used with legumes. Only limited quantities of <sup>15</sup>N can be absorbed by the plant material because of short exposure time. Longer periods of exposure would require sophisticated and expensive equipment unsuitable for field measurements (Bazot et al., 2008).

Urea is a good <sup>15</sup>N carrier because it is non-polar, highly mobile and has a higher N content than nitrate and ammonium. Leaf-feeding (or leaf-flap) methods involve immersing a part of the foliage in a <sup>15</sup>Nlabelled solution contained in a sealed vial for several hours. These have been found to be more accurate than spray applications of <sup>15</sup>N-labelled urea, because of the loss of <sup>15</sup>NH<sub>3</sub> occurring after <sup>15</sup>N-urea hydrolysis and runoff from the labelled leaves to the soil in the case of spray applications (Russell and Fillery, 1996a; Hertenberger and Wanek, 2004). After a spray application of <sup>15</sup>N-urea, Zebarth et al. (1991) recovered less than 70% of the 15N applied in the case of alfalfa and 30% in the case of red clover. Using the leaf-feeding technique, Ledgard et al. (1985) labelled pasture legumes by immersing a trifoliolate leaf into a glass vial sealed in a plastic bag and filled with 15 mL of a 10% <sup>15</sup>N KNO<sub>3</sub> solution (300 mM) for 72 h and measuring N transfers between neighbouring pasture plants. McNeill et al. (1997, 1998) adapted the leaf-feeding technique (Oghoghorie and Pate, 1972; Pate, 1973), to assess below-ground N of subterraneum clover and serradella (Ornithopus compressus L.). They conducted similar experiments in the field and in the greenhouse (McNeill et al., 1997, 1998). After cutting (1997) or not (1998) the 1–2 mm tip, a young expanded leaf was inserted into a 2mL non-porous vial filled with 1 mL of a 0.25-0.4% (w/v) solution of  ${}^{15}$ N-labelled urea (99.6 atom%  ${}^{15}$ N). The system was sealed with inert plastic putty to avoid <sup>15</sup>N loss. To avoid leaf damage, the concentration of the urea solution must not be too high. In the field, mean total recovery of the fed <sup>15</sup>N in the entire plant-soil system at the late vegetative stage was 85% for subterranean clover and 76% for serradella, but was more than 92% in both species after feeding at maturity (Table 1). In the greenhouse, mean recovery of the fed <sup>15</sup>N was 42% in subterranean clover and 64% for serradella. In leaf-feeding methods, <sup>15</sup>N

 
 Table 1
 N rhizodeposited by various legume species as a percentage of the plant N. Values obtained using different labelling methods. (\* injected into labelling compartment soil at the be ginning of the experiment, \*\* continuous labelling in hydroponic compartment, \*\*\* injected every two days in vermiculite of labelling compartment, (fr) including fine roots)

		Culture		<sup>15</sup> N	Rhizodeposited
Reference	Species	conditions	Method	recovery	N / plant-N
Zebarth et al. (1991)	Trifolium pratensis Medicago sativa	Field	Leaf spray	-	-
Sawatsky and Soper	Pisum sativum	Growth	Split-root	_	8-12%
(1991)		chamber	( <sup>15</sup> NH <sub>4</sub> ) <sub>2</sub> SO <sub>4</sub> *		
Jensen (1996a, b)	Pisum sativum	Growth chamber	Split-root KN03-15N **	_	7%
Russell and Fillery (1996b)	Lupinus angustifolius	Field (mesotrons) under rain shelter	Cotton-wick ( <sup>15</sup> N-urea)	81-102%	18.5%
McNeill et al. (1997)	Trifolium subterraneum Ornithopus compressus	Field (mesotrons)	Leaf feeding ( <sup>15</sup> N-urea)	85% 76%	10% (fr) 20% (fr)
McNeill et al. (1998)	Trifolium subterraneum Ornithopus compressus	Greenhouse (pots)	Leaf feeding ( <sup>15</sup> N-urea)	42% 64%	40% (fr) 57% (fr)
Rochester et al. (1998)	Vicia faba ssp minor, Glycine max, Lens culinaris, Lupinus angustifolius, Vigna radiata, V. angularis, V. unguiculata, Cajanus cajan, Arachis hypogaea, Lablab purpureus, Pisum sativum	Field	Petiole feeding ( <sup>15</sup> N-urea)	_	_
Khan et al. (2002a, b)	Vicia faba Cicer arietinum, Vigna radiata, Cajanus cajan	Greenhouse (pots)	Shoot feeding ( <sup>15</sup> N-urea)	90% 76% 100% 102%	23.5% 43.9% 16.5% 35.5%
Chalk et al. (2002)	Sesbania rostrata	Greenhouse (pots)	Leaf feeding ( <sup>15</sup> N-urea) Stem injection ( <sup>15</sup> N-urea) Adventitious root feeding ( <sup>15</sup> N-urea)	35% 45% 101%	- - -
Mayer et al. (2003)	Vicia faba	Cover hall	Cotton-wick	84.8%	13%
	Pisum sativum	(pots)	( <sup>15</sup> N-urea)	83.2%	12%
	Lupinus albus	ч <i>/</i>	· · · ·	84.5%	16%
Schmidtke (2005a, b)	Pisum sativum	Greenhouse	Split-root	_	10.5%
	Lathyrus sativus		KN0 <sub>3</sub> - <sup>15</sup> N ***		9.2%
Yasmin et al. (2006)	Cicer arietinum	Greenhouse (pots)	Leaf feeding Petiole feeding		-
			Cotton-wick		-
Mahieu et al. (2007)	Pisum sativum	Greenhouse (pots)	Cotton-wick ( <sup>15</sup> N-urea)	65-85%	9.7–11.7%
		Field		70%	34.2%
	Pisum sativum	Greenhouse (pots) Field	Split-root <sup>15</sup> NO <sub>3</sub> - <sup>15</sup> NH <sub>4</sub> **	_	14.3–17.3% 27.5%
Gylfadóttir et al.	Mixture	Field	Leaf feeding	-	
(2007)	Trifolium repens Poa pratensis	(mesotrons)	( <sup>15</sup> N-urea)		47% 10% (of total N for both species)
Wichern et al. (2007a)	Pisum sativum	Field (mesotrons)	Cotton-wick (15N-urea)	59–77%	32-36%
McNeill and Fillery (2008)	Lupinus angustifolius	Field (mesotrons)	Cotton-wick ( <sup>15</sup> N-urea)	69–76%	35–65% (fr)

enrichment of above-ground parts is often higher than that of below-ground parts (McNeill et al., 1997, 1998; Yasmin et al., 2006). <sup>15</sup>N leaf-feeding techniques used both by Ledgard et al. (1985) and by McNeill et al. (1998) were also used to measure N compounds deposited in the soil by mixtures of common grassland species in the field and N transfer from legumes to the neighbouring non-fixing plant (Bardgett et al., 1999; Høgh-Jensen and Schjoerring, 2001; Ayres et al., 2007; Rasmussen et al., 2007).

The <sup>15</sup>N solution can be fed directly to a *leaf peti*ole. Rochester et al. (1998) attached vials containing <sup>15</sup>N-urea to petioles of eleven different species of grain legume. Khan et al. (2002b) compared the use of leaffeeding and petiole-feeding methods in the field with four different species. They concluded that <sup>15</sup>N-leafflap feeding was best for fababean, mungbean and pigeonpea, but petiole feeding was best for chickpea. The best compromise to enable comparison of results between species was to apply short pulses of labelled urea to the lower third or fourth stem-node using 0.2 mL of 0.5% urea (98 atom% <sup>15</sup>N) at each pulse. Leaf and petiole feeding led to higher <sup>15</sup>N enrichment of above- than below-ground parts in all tested species except in piegeonpea, where shoot enrichment was about 30% lower than root enrichment (Ledgard et al., 1985; Russell and Fillery, 1996a, b; McNeill et al., 1997, 1998; Khan et al., 2002a; Chalk et al., 2002). In leaf and petiole feeding, although the urea was highly enriched in <sup>15</sup>N, the <sup>15</sup>N enrichment of the roots was only between 0.11 and 0.90 atom% <sup>15</sup>N excess (McNeill et al., 1997; Høgh-Jensen and Schjoerring, 2001; Khan et al., 2002a, b).

(iv) The cotton-wick technique was proposed by Russell and Fillery (1996a). <sup>15</sup>N-labelling solution is provided to the plant by means of a cotton-wick passing through a hole in the plant stem (Fig. 1). These authors have shown that the transfer of solutions into young lupin plants is more effective using the cottonwick method than the leaf-feeding method. N uptake by the cotton-wick technique is mainly driven by the transpiration stream, avoiding active mechanisms occurring with root or leaf immersion. Results reported by Russell and Fillery (1996b) and McNeill and Fillery (2008) confirm that this method seems accurate for assessing below-ground N of field-grown lupin and provides a more homogeneous <sup>15</sup>N distribution in the plants compared with leaf-feeding techniques (Mayer et al., 2003). It has also been confirmed for fababean, chickpea, mungpea (Vigna radiata (L.) R. Wilcz), pigeonpea, pea and white lupin (Russell and Fillery, 1996b; Mayer et al., 2003; Mahieu et al., 2007). Fortnightly pulses of high <sup>15</sup>N-urea (99 atom% <sup>15</sup>N), were found to be more efficient than a weekly application (Russell and Fillery, 1996a) and provide similar results to pulses applied at given growing stages (6-leaf stage, flowering and pod-filling; Mahieu et al., 2007). In Mayer et al. (2003) the amount of urea applied to pea plants at each pulse was calculated from dilution curves, to keep an average <sup>15</sup>N content of 2.5 atom% <sup>15</sup>N excess of the plant N during the growing demand. All experiments undertaken on pea showed that <sup>15</sup>N recovery was around 90% (84-94%) in the greenhouse and 50-76% in the field (Table 1; Mayer et al., 2003; Mahieu et al., 2007; Wichern et al., 2007a). Furthermore, the longer the experiment, the lower <sup>15</sup>N recovery in the plant-soil system (Russell and Fillery, 1996a; Mayer et al., 2003; Mahieu et al., 2007). In cottonwick, as in leaf-flap and petiole feeding, above-ground parts are markedly more <sup>15</sup>N-enriched than roots. Root enrichment ranged between 1.1 and 1.4 atom% <sup>15</sup>N excess in Russell and Fillery (1996a), Mayer et al. (2003) and Wichern et al. (2007a) but reached up to 3.6 atom% <sup>15</sup>N excess in Mahieu et al. (2007). However, cotton-wick cannot be used with thin-stemmed species such as chickpea (Yasmin et al., 2006). Few attempts have been made to inject <sup>15</sup>N-urea directly into the stem with a syringe. Chalk et al. (2002) did not obtain reliable results with S. rostrata, probably because of its hollow stem.

(vi) The split-root technique was proposed by Sawatsky and Soper (1991) to quantify the amount of N lost from the root system of pea plants. Before the beginning of the experiments, seedlings of pea were raised in moist sand or vermiculite, and the radicle was cut 1 cm below the seed after seedling emergence to enhance the development of adventitious roots. Then the root system was split between two different soil compartments. One of them, filled with soil (Sawatsky and Soper, 1991), with clay marbles (Jensen, 1996a, b; Mahieu et al., 2007) or vermiculite (Schmidtke, 2005a, b) was labelled with a mineral <sup>15</sup>N-enriched source, and the other part of the root system growing in the unlabelled soil compartment could be monitored (Fig. 2). Sawatsky and Soper (1991) used a solution of (<sup>15</sup>NH<sub>4</sub>)<sub>2</sub>SO<sub>4</sub> containing 66.7% <sup>15</sup>N: root <sup>15</sup>N enrichment was 9.92 atom% 15N excess. Jensen (1996a), Schmidtke (2005a) and Mahieu et al. (2007)

used a 5% or 10% <sup>15</sup>N-enriched KNO<sub>3</sub>-N, and 10% <sup>15</sup>N-enriched NO<sub>3</sub>-NH<sub>4</sub>, respectively; root enrichments comprised between 0.2 and 3.5% atom% <sup>15</sup>N excess.

This technique can also be used to assess N transfer between a legume and a non-fixing species (Jensen, 1996b). It allows continuous labelling during plant growth and N uptake follows a natural pathway. A disadvantage of the split-root method is that it substantially disturbs the root system and plant development, particularly for species with a taproot (McNeill et al., 1997; Khan et al., 2002a). In addition, roots of the labelling compartment tend to keep more than 50% of the applied <sup>15</sup>N (Schmidtke, 2005b; Mahieu et al., 2007), leading to lower enrichment in N of the other plant parts. Furthermore, estimation of N derived from rhizodeposition accounts for only a part of the root system. This technique is difficult to adapt to field conditions (Mahieu et al., 2007).

# 3.3 Amounts of N Rhizodeposited by Legumes

Among all reviewed studies, N derived from rhizodeposition as a percentage of the mature plant N varied from 7% to 57% (Table 1). Using leaf feeding with <sup>15</sup>N-urea, the ratio of rhizodeposited N: plant N differed markedly among species (from 10% in subterraneum clover to 57% in serradella); values obtained in subterraneum clover and serradella were markedly higher in the field than in the greenhouse (McNeill et al., 1997, 1998).

Several studies have investigated N rhizodeposition of mature pea crops using split-root or cotton-wick methods (Sawatsky and Soper, 1991; Jensen 1996a, b; Mayer et al., 2003; Schmidtke 2005a, b; Mahieu et al., 2007; Wichern et al., 2007a). Harvesting at different stages indicates that N rhizodeposition increases as plants mature, probably because of the increase in senescing roots and nodules (see Wichern et al., 2008). However, Wichern et al. (2007b) measured high levels of rhizodeposition at early vegetative stages of growth (71% of the plant N at the 3–6 leaf stage). For a pea plant, the ratio of rhizodeposited N: plant N was 4 to 71% and the ratio of the below-ground N: plant N varied from 14 to 74%. At maturity, in greenhouse conditions, rhizodeposited N and below-ground N often represented around 15% and 25% of plant N, respectively (Mahieu et al., 2007; see Wichern et al., 2008). In the field, below-ground N represented around 30% of plant N and rhizodeposited N often accounted for 88-97% of below-ground N. Mahieu et al. (2007) showed that the ratio of rhizodeposited N: plant N obtained with split-root was 10% higher than that obtained with cotton-wick. Furthermore, the values were higher in the field than in the greenhouse experiments, though the root-to-shoot ratios were markedly lower in the field than in the greenhouse. Consistently with other studies, roots represented less than 5% of the total plant weight in the field (Voisin et al., 2002), since they represented at least 10-20% of the plant weight in the greenhouse pots. In their greenhouse study, Mahieu et al. (2007) found a significant relationship between the amount of N rhizodeposited by a pea plant and the plant-N content that could contribute to explain this difference, since plant-N contents of field peas were higher than those of greenhouse plants.

## 4 N Rhizodeposition Pathways

## 4.1 General Considerations

The term rhizodeposition was first used to describe carbon loss from roots (Lynch and Wipps, 1990) but also includes N loss, as most organic compounds lost by roots also contain N (but see Wichern et al., 2008). Less N than C is rhizodeposited, but deposition of both elements cannot be distinguished (Bais et al., 2006) as in both cases, the potential pathways for rhizodeposition are (1) senescence, death and decay of roots and nodules; (2) exudation of soluble compounds; (3) sloughing-off of root border cells, and (4) secretion of mucilage. Quantitative data providing reliable estimation of these pathways are sparse but a recent review concerning carbon rhizodeposition showed that sloughing-off of border cells and secretion of mucilage represent a very small proportion of carbon rhizodeposition (N'guyen, 2003). This proportion must be even smaller for N rhizodeposition, as little N is present in mucilages.

#### 4.2 Senescence of Roots and Nodules

Several studies have demonstrated that death of nodules and roots is a major source of biological fixation of N for the soil (Dubach and Russelle, 1994; Russelle et al., 1994). Its importance is undisputed but reliable quantitative data are sparse, as no methodology is available to clearly distinguish rhizodeposition due to death and decay of below-ground tissues from rhizodeposition due to exudation of soluble compounds. By comparing the accumulation of biologically fixed <sup>15</sup>N<sub>2</sub> in fine roots and nodules of alfalfa and birdsfoot trefoil and with soil surrounding the roots, Dubach and Russelle (1994) have estimated that decomposition of these tissues is the main pathway for N rhizodeposition. Though little quantitative data are available concerning fixed nitrogen in legume rhizospheres, quantification of underground N transfer from legumes to intercropped grasses is now well documented. Because transfer increases with plant age, it is often proposed that N release from senescence of belowground residues of legumes coupled with grass uptake is the dominant factor in N exchange (Høgh-Jensen and Schjoerring, 1997; Moyer-Henry et al., 2006). Release of N through degradation of above-ground tissues is highly dependent on numerous factors such as mycorrhizal fungi, bacteria, root herbivory or defoliation (Ta and Faris, 1988; Johansen and Jensen, 1996; Ayres et al., 2007). Numerous studies have established that N transfer between plants can also occur between young plants, through mycorrhizal networks interconnecting plants or through exudation of N compounds by legumes coupled with uptake by grasses (Paynel et al., 2001; Moyer-Henry et al., 2006).

### 4.3 Exudation of Soluble Compounds

The N released from roots and nodules as lowmolecular-weight substances, such as soluble root exudates, amino acids, hormones and enzymes, is also poorly quantified in soil conditions. Most of the experiments on N exudation have been carried out in the laboratory, in hydroponically grown plants or in sand cultures. Between 3% (Ta et al., 1986) and 4.5% of the fixed N is released by alfalfa to the solution as soluble compounds, while between 10 (Brophy and Heichel, 1989) and 30% (Ofosu-Budu, 1990) of fixed N is released to the nutrient solution. *Robinia* pseudoaccacia L., which is an  $N_2$ -fixing tree, has also been observed to release a significant, but minor, proportion of fixed N to the solution, as dissolved organic nitrogen (Uselman et al., 1999).

The main N compound released is generally ammonium, which is the main product of the nitrogenase enzyme, but significant proportions of ureides and amino acids are also recovered in root exudates of alfalfa, soybean and clover (Ta et al., 1986; Brophy and Heichel, 1989; Paynel and Cliquet, 12003). Among amino acids found in root exudates of various species including white clover and alfalfa, glycine and serine have often been recovered in high proportions (Svenningsson et al., 1990; Paynel et al., 2001; Hertenberger and Wanek, 2004; Lesuffleur et al., 2007) despite also constituting a major amino acid in rhizospheric soils (Kielland, 1995; Jones et al., 2005). The reverse is true for other amino acids such as asparagine and glutamine, which are recovered in low proportions in exudates but in high proportions in root extracts, showing that amino acid root exudation is a selective process. Ammonium and amino acids are also recovered in root exudates of non-fixing plants (Paynel and Cliquet, 12003), but use of <sup>15</sup>N-labelled amino acids has shown that efflux of glycine and serine from roots of legumes is higher than from roots of grasses (Lesuffleur et al., 2007). Like the other components of rhizodeposition, root exudation is altered by numerous biotic factors, such as mycorrhizal fungi and root herbivores (Murray et al., 1996; Bais et al., 2006) and abiotic factors, such as defoliation and CO<sub>2</sub> enrichment (Ayres et al., 2007; Bazot et al., 2008).

#### **5** Conclusion

In conclusion, biological fixation of N can act as a sustainable source of N and can complement or replace fertiliser inputs. This review highlights that numerous agricultural practices have been developed all around the world to take advantage of the biological reduction of atmospheric N to ammonia realised by some prokariots. N fixation is performed by these prokariots alone or in symbiosis with plants. Legumes form a symbiosis with *Rhizobium* but release a substantial part of the biologically fixed N into the rhizosphere. As a consequence, biological N fixation can act as a

sustainable source of N and contribute to decreasing fertiliser inputs. However, the part of this N available for non-fixing crops remains difficult to assess. N rhizodeposition is mainly due to senescence and decay of roots and nodules, and exudation of N compounds by living roots. The main N compounds released by legume roots are ammonium, amino acids and ureides, but a wide range of organic compounds released by plant roots remain to be determined. A significant effort has been made in the last decade to develop tracer methods suitable for quantifying N rhizodeposition in realistic conditions. Long-term studies using the splitroot and the cotton-wick techniques have shown that N rhizodeposition increases with plant age and plant N content, but more information is lacking concerning the effects of plant-N partitioning and of root characteristics. Ecological functions of these rhizodeposits are still unknown, but they may constitute a rapidly incorporating source of C and N for soil microorganisms and neighbouring plants. Further investigations combining assessments of C and N rhizodeposition are needed to obtain a better understanding of these fluxes in the rhizosphere of legumes.

#### References

- Ayres E., Dromph K.M., Cook R., Ostle N., Bardgett R.D. (2007) The influence of below-ground herbivory and defoliation of a legume on nitrogen transfer to neighbouring plants, Funct. Ecol. 21, 256–263.
- Bais H.P., Weir T.L., Perry L.G., Gilroy S., Vivanco J.M. (2006) The role of root exudates in rhizosphere interactions with plants and other organisms, Ann. Rev. Plant Biol. 57, 233–266.
- Bardgett R.D., Denton C.S., Cook R. (1999) Below ground herbivory promotes soil nutrient transfer and root growth in grassland, Ecol. Lett. 2, 357–360.
- Bazot S., Blum H., Robin C. (2008) Nitrogen rhizodeposition assessed by a <sup>15</sup>NH<sub>3</sub> shoot pulse-labelling of *Lolium perenne* L. grown on soil exposed to 9 years of CO<sub>2</sub> enrichment, Environ. Exp. Bot. 63, 410–415.
- Bergersen F.J., Brockwell J., Gault R.R., Morthorpe L., Peoples M.B., Turner G.L. (1989) Effects of available soil nitrogen and rates of inoculation on nitrogen fixation by irrigated soybeans and evaluation of  $\delta^{15}$ N methods for measurements, Aust. J. Agr. Resour. Ec. 40, 763–780.
- Brophy L.S., Heichel G.H. (1989) Nitrogen release from roots of alfalfa and soybean grown in sand culture, Plant Soil 116, 77–84.
- Burris R.H. (1974) Biological nitrogen fixation, Plant Physiol. 54, 443–449.
- Carlsson G., Huss-Danell K. (2003) Nitrogen fixation in perennial forage legumes in the field, Plant Soil 253, 353–372.

- Chapman A.L., Myers R.J.K. (1987) Nitrogen contribution by grain legumes to rice grown in rotation on the Cununurra soils the Ord irrigation area, Western Australia, Aust. J. Agr. Resour. Ec. 48, 1139–1150.
- Chalk P.M. (1998) Dynamics of biologically fixed N in legumecereal rotations: a review, Aust. J. Agr. Resour. Ec. 49, 303–316.
- Chalk P.P., Ladha J.K., Padre A. (2002) Efficacy of three <sup>15</sup>N labelling techniques for estimating below ground N in Sesbania rostrata, Biol. Fert. Soils 35, 387–398.
- Corre-Hellou G., Fustec J., Crozat Y. (2006) Interspecific competition for soil N and its interaction with N<sub>2</sub> fixation, leaf expansion and crop growth in pea-barley intercrops, Plant Soil 282, 195–208.
- Corre-Hellou G., Brisson N., Launay M., Fustec J., Crozat Y. (2007) Effect of root depth penetration on soil N sharing and dry matter in pea-barley intercrops given different soil N supplies, Field Crop. Res. 103, 76–85.
- Crawford M.C., Grace P.R., Bellotti W.D., Oades J.M. (1997) Root production of a barrel medic (*Medicago truncatula*) pasture, a barley grass (*Hordeum leporinum*) pasture, and a fababean (*Vicia faba*) crop in Southern Australia, Aust. J. Agr. Resour. Ec. 48, 1139–1150.
- Crépon K. (2006) Protein supply in Europe and the challenge to increase grain legumes production: a contribution to sustainable agriculture, in: Grain legumes and the environment: how to assess benefits and impacts. Proceedings of the AEP workshop, (Ed.) AEP, 18–19 November 2004, Zürich, Switzerland, pp. 13–16.
- Crozat Y., Fustec J. (2006) Assessing the role of grain legumes in crop rotation: some agronomic concepts that can help!, in: Grain legumes and the environment: how to assess benefits and impacts. Proceedings of the AEP workshop, (Ed.) AEP, 18–19 November 2004, Zürich, Switzerland, pp. 55–60.
- Curatti L., Ludden P.W., Rubio L.M. (2006) NifB-dependent in vitro synthesis of the iron-molybdenum cofactor of nitrogenase, PNAS 103, 5297–5301
- Dalal R.C., Strong W.M., Doughton J.A., Weston E.J., McNamara G.T., Cooper J.R. (1997) Sustaining productivity of a vertisol at warra, Queensland, with fertilizer, notillage or legumes. 4. Nitrogen fixation, water use and yield of chickpea, Aust. J. Exp. Agr. 37, 667–676.
- Deutsch B., Kahle P., Voss M. (2006) Assessing the source of nitrate pollution in water using stable N and O isotopes, Agron. Sustain. Dev. 26, 263–267.
- Dubach M., Russelle M.P. (1994) Forage legume roots and nodules and their role in nitrogen transfer, Agron. J. 86, 259–266.
- Frame J., Charlton J.F.L., Laidlaw A.S. (1998) Temperate forage legumes, CAB International, Wallingford, UK, 327 p.
- Fujita K., Ofosu-Budu K.G., Ogata S. (1992) Biological nitrogen fixation in mixed legume-cereal cropping systems, Plant Soil 141, 155–175.
- Garg N., Geetanjali (2007) Symbiotic nitrogen fixation in legume nodules: process and signaling. A review, Agron. Sustain. Dev. 27, 59–68.
- Giller K.E., Ormesher J., Awah F.M. (1991) Nitrogen transfer from *Phaseolus* bean to intercropped maize measured using <sup>15</sup>N-enrichment and <sup>15</sup>N-isotope dilution methods, Soil Biol. Biochem. 23, 339–346.
- Gordon A.J., Lea P.J., Rosenberg C., Trinchant J.C. (2001) Nodule formation and function, in: Lea P.J., Morot-Gaudry

J.F. (Eds.), Springer-Verlag Plant nitrogen, Berlin Heidelberg New York, pp. 101–146.

- Gylfadóttir T., Helgadóttir A., Høgh-Jensen H. (2007) Consequences of including adapted white clover in northern European grassland: transfer and deposition of nitrogen, Plant Soil 297, 93–104.
- Hansson A.C., Steen E. (1984) Methods of calculating root production and nitrogen uptake in an annual crop, Swedish J. Agr. Res. 14, 191–200.
- Hartwig U.A. (1998) The regulation of symbiotic  $N_2$  fixation: a conceptual model of N feedback from the novel ecosystem to the gene expression level, Perspect. Plant Ecol. Evol. Syst. 1/1, 92–120.
- Hauggaard-Nielsen H., Jensen E.S. (2005) Facilitative root interactions in intercrops, Plant Soil 274, 237–250.
- Hertenbergen G., Wanek W. (2004) Evaluation of methods to measure differential <sup>15</sup>N labeling of soil and root-N pools for studies of root exudation, RCM 18, 2415–2425.
- Hétier J.M., Andreux F., Schouller E., Marol C. (1986) Organic matter inputs after growth of Carbon 14-Nitrogen labeled maize, Soil Sci. Soc. Am. J. 50, 76–80.
- Høgh-Jensen H. (2006) The nitrogen transfer between plants: An important but difficult flux to quantify, Plant Soil 282, 1-5.
- Høgh-Jensen H., Schjoerring J.K. (1997) Interaction between white clover and ryegrass under contrasting nitrogen availability: N<sub>2</sub> fixation, N fertilizer recovery, N transfer and water-use efficiency, Plant Soil 197, 187–199.
- Høgh-Jensen H., Schjoerring J.K. (2001) Rhizodeposition of nitrogen by red clover, white clover and ryegrass leys, Soil. Biol. Biochem. 33, 439–448.
- Janzen H.H., Bruinsma Y. (1989) Methodology for the quantification of root and rhizosphere N dynamics, Soil. Biol. Biochem. 21, 189–196.
- Jensen E.S. (1996a) Rhizodeposition of N by pea and barley and its effect on soil N dynamics, Soil. Biol. Biochem. 28, 65–71.
- Jensen E.S. (1996b) Barley uptake of N deposited in the rhizosphere of associated field pea, Soil. Biol. Biochem. 28, 159–168.
- Jensen E.S. (2006) Grain legume functions in crop rotations, in: Grain legumes and the environment: how to assess benefits and impacts. Proceedings of the AEP workshop, (Ed.) AEP, 18–19 November 2004, Zürich, Switzerland, pp. 49–54.
- Johansen A., Jensen E.S. (1996) Transfer of N and P from intact or decomposing roots of pea to barley interconnected by an arbuscular mycorhizal fungus, Soil Biol. Biochem. 28, 73–81.
- Jones D.L., Healey J.R., Willett V.B., Farrar J.F., Hodge A. (2005) Dissolved organic nitrogen uptake by plants, an important N uptake pathway? Soil Biol Biochem 37, 413–423.
- Khan D.F., Peoples M.B., Chalk P.M., Herridge D.F. (2002a) Quantifying below-ground nitrogen of legumes. 2. A comparison of <sup>15</sup>N and non isotopic methods, Plant Soil 239, 277–289.
- Khan D.F., Peoples M.B., Herridge D.F. (2002b) Quantifying below ground nitrogen of legumes. 1. Optimising procedures for <sup>15</sup>N shoot-labelling, Plant Soil 245, 327–334.
- Khan D.F., Herridge D.F., Peoples M.B., Shah S.H., Khan T., Madani M.S., Ibrar M. (2007) Use of isotopic and

non-isotopic techniques to quantify below-ground nitrogen in fababean and chickpea, Soil Environ. 26, 42–47.

- Kielland K. (1995) Landscape patterns of free amino acids in Arctic tundra soils, Biogeochemistry 31, 85–98.
- Ledgard S.F., Freney J.R., Simpson J.R. (1985) Assessing nitrogen transfer from legumes to associated grasses, Soil. Biol. Biochem. 17, 575–577.
- Lesuffleur F., Paynel F., Bataille M.P., Le Deunff E., Cliquet J.B. (2007) Root amino acid exudation: Measurement of high efflux rates of glycine and serine from six different plant species, Plant Soil 294, 235–246.
- Loreau M., Hector A. (2001) Partitioning selection and complementarity in biodiversity experiments, Nature 412, 72–76.
- Lynch J.M., Wipps T.M. (1990) Substrate flow in the rhizosphere, Plant Soil 129, 1–10.
- Mahieu S., Fustec J., Faure M.L., Corre-Hellou G., Crozat Y. (2007) Comparison of two <sup>15</sup>N labelling methods for assessing nitrogen rhizodeposition of pea, Plant Soil 295, 193–205.
- Mayer J., Buegger F., Jensen E.S., Schloter M., He $\beta$  J. (2003) Residual nitrogen contribution from grain legumes to succeeding wheat and rape and related microbial process, Soil Biol. Biochem. 35, 21–28.
- McNeill A.M., Fillery I.R.P. (2008) Field measurement of lupin below ground nitrogen accumulation and recovery in the subsequent cereal-soil system in a semi-arid Mediterranean-type climate, Plant Soil 302, 297–316.
- McNeill A.M., Hood R.C., Wood M. (1994) Direct measurement of nitrogen fixation by *Trifolium repens* L. and *Alnus glutinosa* L. using <sup>15</sup>N<sub>2</sub>, J. Exp. Bot. 45, 749–755.
- McNeill A.M., Zhu C., Fillery I.R.P. (1997) Use of in situ <sup>15</sup>Nlabelling to estimate the total below-ground nitrogen of pasture legumes in intact soil-plant systems, Aust. J. Agr. Resour. Ec. 48, 295–304.
- McNeill A.M., Zhu C., Fillery I.R.P. (1998) A new approach to quantifying the N benefit from pasture legumes to succeeding wheat, Aust. J Agr. Resour. Ec. 49, 427–436.
- Moyer-Henry K.A., Burton J.W., Israel D.W., Rufty T.W. (2006) Nitrogen Transfer Between Plants: A <sup>15</sup>N Natural Abundance Study with Crop and Weed Species, Plant Soil 282, 7–20.
- Murray P.J., Hatch D.J., Cliquet J.B. (1996) Impact of insect herbivory on the growth, nitrogen and carbon contents of white clover (*Trifolium repens* L.), Can. J. Bot. 74, 1591–1595.
- N'guyen C. (2003) Rhizodeposition of organic C by plant: mechanisms and controls, Agronomie 23, 375–396.
- Ofosu-Budu K.G., Fujita K., Ogata S. (1990) Excretion of ureide and other nitrogenous compounds by the root system of soybean at different growth stages, Plant Soil 128, 135–142.
- Oghoghorie C.G.O., Pate J.S. (1972) Exploration of the nitrogen transport system in a nodulated legume using <sup>15</sup>N, Planta 104, 35–49.
- Padilla F.M., Pugnaire F.I. (2006) The role of nurse plants in restoration of degraded environments, Front. Ecol. Environ. 4, 196–202.
- Pate J.S. (1973) Uptake, assimilation and transport of nitrogen compounds by plants, Soil Biol. Biochem. 5, 109–119.
- Paynel F., Cliquet J.B. (2003) <sup>15</sup>N transfer from white clover to perennial ryegrass, via exudation of nitrogenous compounds, Agronomie 23, 503–510.

- Paynel F., Murray P., Cliquet J.B. (2001) Root exudates: a pathway for short-term N transfer from clover and ryegrass, Plant Soil 229, 235–243.
- Paynel F., Lesuffleur F., Bigot J., Diquélou S., Cliquet J.B. (2008) A study of <sup>15</sup>N transfer between legumes and grasses, Agron. Sustain. Dev. 28, 281–290.
- Poth M., La Favre J.S., Focht D.D. (1986) Quantification by direct <sup>15</sup>N dilution of fixed N<sub>2</sub> incorporation into soil by *Cajanus cajan* (pigeon pea), Soil Biol. Biochem. 18, 125–127.
- Rasmunssen J., Eriksen J., Jensen E.S., Esbensen K.H., Høgh-Jensen H. (2007) *In situ* carbon and nitrogen dynamics in rye-grass clover mixtures: transfers, deposition and leaching, Soil Biol. Biochem. 39, 804–815.
- Rochester I.J., Peoples M.B., Constable G.A., Gault R.R. (1998) Fababean and other legumes add nitrogen to irrigated cotton cropping systems, Aust. J. Exp. Agr. 38, 253–260.
- Rochon J.J., Doyle C.J., Greef J.M., Hopkins A., Molle G., Sitzia M., Scholefield D., Smith C.J. (2004) Grazing legumes in Europe: a review of their status, management, benefits, research needs and future prospects, Grass Forage Sci. 59, 197–214.
- Russell C.A., Fillery I.R.P. (1996a) In situ <sup>15</sup>N labelling of lupin below ground biomass, Aust. J. Agr. Resour. Ec. 47, 1035–1046.
- Russell C.A., Fillery I.R.P. (1996b) Estimates of lupin below ground biomass nitrogen, dry matter, and nitrogen turnover to wheat, Aust. J. Agr. Resour. Ec. 47, 1047–1059.
- Russelle M.P., Allan D.L., Gourley C.J.P. (1994) Direct assessment of symbiotically fixed nitrogen in the rhizosphere of alfalfa, Plant Soil 159, 233–243.
- Sawatsky N., Soper R.J. (1991) A quantitative measurement of the nitrogen loss from the root system of field peas (*Pisum avense* L.) grown in the soil, Soil. Biol. Biochem. 23, 255–259.
- Schmidtke K. (2005a) A model to predict the accuracy of measurements of legume N rhizodeposition using a split-root technique, Soil Biol. Biochem. 37, 829–836.
- Schmidtke K. (2005b) How to calculate nitrogen rhizodeposition: a case study in estimating N rhizodeposition in the pea (*Pisum sativum L.*) and grasspea (*Lathyrus sativus L.*) using a continuous <sup>15</sup>N labelling split-root technique, Soil Biol. Biochem. 37, 1893–1897.
- Soussana J.F., Schmidtke O. (2000) Modelling the temperate grasses and legumes in cut mixtures, in: Grassland Ecophysiology and Grazing Ecology, Lemaire et al. (Eds.), CAB International London, UK, pp. 169–190.
- Svenningsson H., Sundin P., Liljenberg C. (1990) Lipids, carbohydrates and amino acids exuded from the axenic roots of rape seedlings exposed to water-deficit stress, Plant Cell Environ. 13, 155–162.

- Ta T.C., Faris M.A. (1988) Effects of environmental conditions on the fixation and transfer of nitrogen from alfalfa to associated timothy, Plant Soil 107, 25–30.
- Ta T.C., Macdowall F.D.H., Faris M.A. (1986) Excretion of nitrogen assimilated from N<sub>2</sub> fixed by nodulated roots of alfalfa (*Medicago sativa*), Can. J. Bot. 64, 2063–2067.
- Toomsan B., McDonagh J.F., Limpinuntana V.J.H.A., Giller K.E. (1995) Nitrogen fixation by groundnut and soyabean and residual nitrogen benefits to rice in farmers' fields in Northeast Thailand, Plant Soil 175, 45–56.
- Umar A.S., Iqbal M. (2007) Nitrate accumulation in plants, factors affecting the process, and human health implications. A review, Agron. Sustain. Dev. 27, 45–57.
- Unkovich M.J., Pate J.S. (2000) An appraisal of recent field measurements of symbiotic N<sub>2</sub> fixation by annual legumes, Field Crop. Res. 65, 211–228.
- Uselman S.M., Qualls R.G., Thomas R.B. (1999) A test of a potential short cut in the nitrogen cycle: The role of exudation of symbiotically fixed nitrogen from the roots of a N-fixing tree and the effects of increased atmospheric CO<sub>2</sub> and temperature, Plant Soil 210, 21–32.
- Vance C.P. (2001) Symbiotic nitrogen fixation and phosphorus acquisition: plant nutrition in a world of declining renewable resources, Plant Physiol. 127, 390–397.
- Voisin A.S., Salon C., Munier-Jolain N.G., Ney B. (2002) Effect of mineral nitrogen nutrition and biomass partitioning between the shoot and roots of pea (*Pisum sativum* L.), Plant Soil 242, 251–262.
- Wichern F., Mayer J., Joergensen R.G., Müller T. (2007a) Rhizodeposition of C and N in peas and oats after <sup>13</sup>C-<sup>15</sup>N double labelling under field conditions, Soil Biol. Biochem. 30, 2527–2537.
- Wichern F., Mayer J., Joergensen R.G., Müller T. (2007b) Release of C and N from roots of peas and oats and their availability to soil microorganisms, Soil Biol. Biochem. 39, 2829–2839.
- Wichern F., Eberhardt E., Mayer J., Joergensen R.G., Müller T. (2008) Nitrogen rhizodeposition in agricultural crops: methods, estimates and future prospects, Soil Biol. Biochem. 40, 30–48.
- Warembourg F.R., Montange D., Bardin R. (1982) The simultaneous use of <sup>14</sup>CO<sub>2</sub> and <sup>15</sup>N<sub>2</sub> labelling techniques to study the carbon and nitrogen economy of legumes grown under natural conditions, Physiol. Plant. 56, 46–55.
- Yasmin K., Cadish G., Baggs E.M. (2006) Comparing <sup>15</sup>Nlabelling techniques for enriching above- and below ground components of the plant-soil system, Soil. Biol. Biochem. 38, 397–400.
- Zebarth B.J., Alder V., Sheard R.W. (1991) *In situ* labeling of legume residues with a foliar application of a [15]Nenriched urea solution, Commun. Soil Sci. Plant Anal. 22, 437–447.

# **Models of Biological Nitrogen Fixation of Legumes**

Yanyan Liu, Lianhai Wu, John A. Baddeley, and Christine A. Watson

#### Contents

1	Intro	duction	884		
2	Facto	rs Influencing Legume Biological			
	Nitro	gen Fixation	885		
	2.1	Temperature	885		
	2.2	Soil Water Status	886		
	2.3	N Concentration in the Root Zone	887		
	2.4	Carbon Demand for Fixation	887		
	2.5	Seasonal Regulation of Legume Biological			
		Nitrogen Fixation	887		
3	Quan	tification of Legume Biological			
	Nitro	gen Fixation	888		
	3.1	Estimation With Empirical Models	888		
	3.2 Mechanistic Simulation of Legume				
		Biological Nitrogen Fixation as a Subroutine			
		in Crop Models	889		
	3.3	Summary of Approaches	897		
4	Discu	ssion	898		
5	Conc	usion	900		
Re	ferenc	es	900		

Y. Liu (🖂)

Abstract Leguminous crops have the ability to fix nitrogen (N) biologically from the atmosphere. This can benefit not only the legumes themselves but also any intercropped or subsequent crops, thus reducing or removing the need to apply N fertilizers. Improved quantification of legume biological nitrogen fixation (BNF) will provide better guidance for farmers on managing N to optimise productivity and reduce harmful losses to the environment. There are many techniques available for the direct quantitative measurement of legume BNF in the field and in controlled environments. However, these are time-consuming and therefore expensive, and generate data relevant only to the time and place of measurement. Alternatively, legume BNF can be estimated by either empirical models or dynamic mechanistic simulation models. Comparatively, simulation by a dynamic model is preferable for quantifying legume BNF, because of its capability to simulate the response of N fixation to a wide range of environmental variables and legume growth status. Currently there is no published review of the approaches used to simulate, rather than measure, legume BNF. This review of peer-reviewed literature shows that most simulation models estimate the N fixation rate from a pre-defined potential N fixation rate, adjusted by the response functions of soil temperature, soil/plant water status, soil/plant N concentration, plant carbon (C) supply and crop growth stage. Here, we highlight and compare the methods used to estimate the potential N fixation rate, and the response functions to simulate legume BNF, in nine widely-cited models over the last 30 years. We then assess their relative strengths in simulating legume BNF with varying biotic and abiotic factors, and identify the discrepancies between experimental findings and

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_39,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Liu et al., Agron. Sustain. Dev. (2010). In press. DOI: 10.1051/agro:2010008

College of Resources and Environmental Sciences, China Agricultural University, Beijing 100193, China and

Crop and Soil Systems Research Group, Scottish Agricultural College, Craibstone Estate, Aberdeen, AB21 9YA, UK e-mail: yanyan.liu@cau.edu.cn

simulations. After this comparison, we identify the areas where there is the potential to improve legume BNF simulation in the future. These include; (1) consideration of photosynthetic C supply, (2) refining the various effects of soil mineral N concentration, (3) characterization and incorporation of excess soil water stress and other factors into models, and (4) incorporation of the effects of grazing, coexistence and competition with intercrops and weeds into models to improve their practical relevance to sustainable agricultural systems. This review clarifies, for the first time, the current progress in legume BNF quantification in simulation models, and provides guidance for their further development, combining fundamental experimental and modelling work.

**Keywords** Nitrogen fixation • Soil mineral nitrogen• Legume • Simulation • Review

## 1 Introduction

Increasing demand for agricultural production typically results in greater fertilizer application (Whitehead, 1995; Wood, 1996). Nitrogen (N) applied in fertilizers or manures which is not taken up by crops can be released into the atmosphere as nitrogenous greenhouse gases (Flechard et al., 2007) or leached into ground water (Stout et al., 2000; Trindade et al., 2001), with resulting environmental implications. Rather than relying purely on applications of N fertilizer, alternative N sources are needed to help develop more sustainable farming systems. Legumes have the potential to fulfil this requirement due to their unique ability to fix N biologically from the atmosphere, benefiting not only the legumes themselves but also the intercropped or subsequent crops.

The ability to fix atmospheric N comes from the symbiotic relationship between legumes and rhizobia, bacteria in soil, through which the legume supplies energy and carbon (C) to rhizobia through the products of photosynthesis, and rhizobia provides the legume with N, mainly in the form of ammonium (Howard and Rees, 1996). The symbiosis initiates from the infection of legume roots by *Rhizobium*, forming root nodules where N fixation occurs. The N fixation process needs

the nitrogenase enzyme, as a catalyst for the reaction to split the nitrogen molecule and change it into ammonia with energy and electron supply (Postgate, 1982). Garg and Geetanjali (2007) reviewed the processes and signalling involved in symbiotic N fixation in legume nodules at a micro scale.

This kind of symbiosis exists in many types of legumes, including grain legumes, forage legumes, and some leguminous trees. Grain legumes and forage legumes are used to build up soil N fertility in sustainable farming systems (Hossain et al., 1996; Jensen, 1997; McCallum et al., 1999), especially in temperate zones. The benefit to soil N fertility from grain/forage legumes is positively correlated with N fixation (Evans et al. 1989, 2001) assuming that the plant residues are incorporated into soil after crop harvest. Therefore, quantifying legume biological N fixation (BNF) is very important for a better understanding of how to design sustainable farming systems, where subsequent crops benefit from legume N fixation, and where N losses and thus environmental pollution may be reduced.

There are many techniques available for measuring legume BNF in the field and in controlled environments (Goh et al., 1978; Sheehy et al., 1991; Herridge et al., 2008; Carlsson and Huss-Danell, 2008). Most of these methods involve destructive sampling of plants or disturb the original soil structure, except for the acetylene (C<sub>2</sub>H<sub>2</sub>) reduction assay (Carlsson and Huss-Danell, 2008). However, the  $C_2H_2$  reduction assay can only be used over a short time period (minutes or hours), and legume N fixation rate varies seasonally with growth stage of the plant, so the time of measurement affects precision and reliability (Carlsson and Huss-Danell, 2008). These methods are able to reveal the response of N fixation to varying factors in real situations, but they are limited to the conditions prevailing at the time of measurement and cannot be used to predict N fixation.

Thus, there is a need for methods to predict legume BNF quantitatively to help make decisions about developing and managing sustainable farming schemes. Estimating N fixation from crop yield or empirical models based on historical crop data can be used, but these are restricted to a particular site for the same crop. Modelling is potentially a better tool to understand and quantify legume BNF as it is based on knowledge of the mechanisms involved, climate and management, and can accurately represent variations in legume BNF under different situations. However, models require very detailed information on the environment and plant genetic performance, in addition to historical crop datasets for calibration and validation. This means that models can not be applied to a given site unless this data is available.

A further complication is that the mathematical functions used to simulate legume BNF vary between models. For example, the rate of potential N fixation is estimated either by plant N demand and uptake (Bouniols et al., 1991; Cabelguenne et al., 1999), nodule biomass (Boote et al., 2002; Wu and McGechan, 1999), root biomass (Thornley et al., 1995; Soussana et al., 2002) or aboveground biomass (Sinclair, 1986; Robertson et al., 2002). Also the response of N fixation to soil mineral N concentration in the root zone, which is closely correlated to N fixation and thus to the benefit to soil N fertility from legumes (Evans et al. 1987, 1989), differs between models. N fixation is either very sensitive (Wu and McGechan, 1999; Schwinning and Parsons, 1996) or tolerant (Sharpley and Williams, 1990; Bouniols et al., 1991) to high soil N concentration in different models.

In this paper we explore and compare the modelling methods published for quantifying legume BNF by grain and forage legumes at the plot/field scale. More specifically, we contrast the different schemes and functions used to simulate the response of N fixation to biotic and abiotic factors and analyse their relative strengths and weaknesses. We then identify gaps in the current models and make suggestions to improve the simulation of legume BNF in future models.

# 2 Factors Influencing Legume Biological Nitrogen Fixation

The complex process of legume BNF is affected by environmental conditions such as temperature, water content, N concentration, root zone pH, plant nutrient status including C and N substrates in roots, and genetic variation in potential N fixation capacity. It is also affected by plant nutritional status such as phosphorus (P) and potassium (K) levels that control nodule growth and nitrogenase activity directly or indirectly (Havelka et al., 1982). In this section we focus only on how legume BNF responds to environmental conditions and plant C and N substrates with a view to improving models.

#### 2.1 Temperature

Generally, soil temperature inhibits legume BNF through its control on nodulation, nodule establishment, and nitrogenase activity when it is either too high or too low (Roughley and Dart, 1970; Roughley, 1970; Whitehead, 1995). Therefore, minimum and maximum soil temperatures, and the range of temperature between these which are favourable for N fixation, could be used to define the response of N fixation to soil temperature.

Soil temperature in the root zone is one of controlling factors for nodulation and nodule establishment. For example, the nodulation of arrowleaf clover (Trifolium vesiculosum Savi.) is accelerated at a root temperature of 25°C compared with that growing at both 18°C and 32°C (Schomberg and Weaver, 1992). However, the response of nodule establishment to soil temperature differs between species and varieties. In soybean (Glycine max (L.) Merr.), more nodules are produced in the early growth stage at 25°C, while 20°C is optimal for nodule size after nodule generation is completed compared to 15°C and 30°C (Lindemann and Ham, 1979). In contrast, nodule establishment is enhanced with increasing temperature in the range of 10-35°C for white clover (Trifolium repens L.) regardless of the varieties and the rhizobia strains (Richardson and Syers, 1985; Whitehead, 1995).

Nitrogenase activity responds slightly differently to soil temperature between species. There are a large number of studies on the response of N fixation to temperature in legume crops, some of which are summarized in Table 1. Minimum temperatures for N fixation differ among species from 2°C to 10°C, and normally tropical and subtropical legumes have higher minimum temperatures than temperate species. Nitrogenase activity is high around 12–35°C and reaches maximum at 20–25°C in most legumes. N fixation in subterranean clover (*Trifolium subterraneum* L.) is very active at a wide range of temperatures, from 5°C to 30°C, but declines dramatically with low temperature and almost stops at 2°C. Generally, 35°C or 40°C is the upper limit of temperature for legume BNF.

Species	Minimum	Optimum range	Maximum	Reference
Alfalfa (Medicago sativa L. )	2	20-25		Waughman (1977)
	2	35	40	Dart and Day (1971)
Arrowleaf clover (Trifolium vesiculosum)		25		Schomberg and Weaver (1992)
Barrel medic (Medicago truncatula Gaertn.)	2	20	35	Dart and Day (1971)
Big-leaved lupin (Lupinus polyphyllus Lindl.)	1.5	25		Waughman (1977)
Birdsfoot trefoil (Lotus corniculatus L.)	10	25-27	35	Rao (1977)
			35	Piha and Munns (1987)
Common bean (Phaseolus vulgaris L.)			40	Michiels et al. (1994)
Common vetch (Vicia sativa L.)	2	20	40	Dart and Day (1971)
Cowpea (Vigna unguiculata (L.) Walp)	5	40		Dart and Day (1971)
	2	20-30	40	Dart and Day (1971)
Faba bean ( <i>vicia faba</i> L.)	5	20		Waughman (1977)
	5	15-25		Halliday (1975)
Field/garden pea (Pisum sativum L.)	0.5	25		Waughman (1977)
		20-26		Lie (1971)
Guar (Cyamopsis tetragonoloba Guar)			37-40	Pate (1961)
Narrowleaf lupin (Lupinus angustifolius L.)	10	20-30		Halliday (1975)
Peanut (Arachis hypogaea L.)			40	Kishinevsky et al. (1992)
Purple vetch (Vicia atropurpurea Desf.)		24		Pate (1961)
	2	25	35	Dart and Day (1971)
Red clover (Irifolium pratense L.)		12-26		Small and Joffe (1968)
		27		Kuo and Boersma (1971)
Soybean (Glycine max L.)		20-25		Lindemann and Ham (1979)
	5	20-35	40	Dart and Day (1971)
	2	5-30		Dart and Day (1971)
Subterranean clover ( <i>Trifolium subterraneum</i> L.)	5	12-32		Gibson (1971)
· · · · · · · · · · · · · · · · · · ·		15		Roughley and Dart (1969)
	5	13-26		Halliday and Pate (1976)
		26		Small and Joffe (1968)
White clover (Trifolium repens L.)		21		Masterson and Murphy (1976)
	9			Frame and Newbould (1986)

Table 1 Response of N fixation to soil temperature (°C) for legume species

#### 2.2 Soil Water Status

In a similar manner to soil temperature, soil water content in the root zone controls N fixation through nodule establishment and nodule activity, plus gas permeability (Sprent, 1976; Weisz et al., 1985; Weisz and Sinclair, 1987; Sinclair et al., 1987). Soil water deficit inhibits N fixation (Whitehead, 1995; Goh and Bruce, 2005), and the inhibition is reinforced as drought stress becomes more intense (Albrecht et al., 1984). N fixation by peanut (*Arachis hypogaea* L.) grown in soil at two-thirds of field water capacity is reduced by 18–40% compared to that in well-watered soil, and by 44–69% when the soil water is one-third of field capacity (Pimratch et al., 2008). In addition, waterlogging can seriously reduce N fixation through depression of the establishment and activity of nodules (Havelka et al., 1982).

It is not easy to quantify the relationship between N fixation and soil water status precisely, due to the limitations of experimental controls, stress periods and plant recovery (Engin and Sprent, 1973; Ledgard and Steele, 1992). Pimratch et al. (2008) tried to determine N fixation under drought stress with multiple-linear regression but the correlation coefficients between the reduction of N fixation under drought compared to under field capacity and the soil water deficit was not statistically significant. A sigmoid function has been used to describe the response of N fixation by common bean (*Phaseolus vulgaris* L.), black gram (*Vigna mungo* (L.)

Hepper) and cowpea (*Vigna unguiculata* (L.) Walp) to soil water stress, showing a sharper decline in N fixation as soil water stress becomes more severe (Serraj and Sinclair, 1998; Sinclair et al., 1987).

#### 2.3 N Concentration in the Root Zone

It has been widely reported that soil mineral N in the root zone inhibits legume nodulation (Abdel Wahab et al., 1996; Herridge et al., 1984), nodule establishment (Imsande, 1986) and nitrogenase activity (Purcell and Sinclair, 1990; Eaglesham, 1989) as it costs less energy for legumes to take up N from soil than fix N biologically from the atmosphere (Cannell and Thornley, 2000; Phillips, 1980; Ryle et al., 1979; Wood, 1996).

Normally, the severity of inhibition of N fixation by soil mineral N increases with soil mineral N content (Macduff et al., 1996; Lamb et al., 1995; Waterer and Vessey, 1993a; Chalifour and Nelson, 1987). It has been reported that a certain concentration of mineral N in the root zone, defined as "starter N", stimulates nodule establishment and N fixation compared to non-mineral N in some circumstances. And the concentrations of "starter N" that stimulate legume BNF vary widely with cultivar and growth conditions but are normally less than 4mM for ammonium  $(NH_4^+)$ and less than 2 mM for nitrate  $(NO_3^-)$  (Bethlenfalvay et al., 1978; Schomberg and Weaver, 1992; Gulden and Vessey, 1997; Gan et al., 2004). However, the time of external N application in relation to legume growth stage affects nodule growth and N fixation, and the later N is applied, the less nodule growth and N fixation is inhibited in field pea (*Pisum sativum* L.) (Jensen, 1986; Waterer and Vessey, 1993b). This is probably due to nodules being well-established in the early stages before N is applied. Moreover, the inhibition of N fixation by  $NO_3^-$  was more severe than that of  $NH_4^+$  in white clover, field pea and soybean (Svenning et al., 1996; Bollman and Vessey, 2006; Gan et al., 2004) even though high amounts of either  $NH_4^+$  or  $NO_3^-$  retarded N fixation.

#### 2.4 Carbon Demand for Fixation

Photosynthate partitioned to roots supports nodule growth, provides energy for N fixation, maintains a functional population of rhizobia, and allows the synthesis of amino compounds produced from N fixation (Minchin and Pate, 1973; Layzell et al., 1979; King et al., 1986). Even though it is difficult to distinguish the proportion of  $CO_2$  generated by N fixation from that generated by respiration for nodule growth and maintenance (Warembourg and Roumet, 1989), the correlation between the rate of  $CO_2$  produced from either nodulated roots or nodules and N fixation rate may be used to evaluate C consumption by N fixation (Lawrie and Wheeler, 1973; Mahon, 1977a, b; Warembourg and Roumet, 1989).

The C cost per unit fixed N (gC per g N fixed) varies widely with species, growth stage and environmental conditions, and ranges from 1.4 to 8.5 (Minchin and Pate, 1973; Phillips, 1980; Minchin et al., 1981; Sheehy, 1987; Schulze et al., 1999). For example, it is reported that the C cost is 1.54 gC g N fixed<sup>-1</sup> in cowpea and 3.64 gC g N fixed<sup>-1</sup> in white lupin (Lupinus albus L.) from nodules during early vegetative stage (Layzell et al., 1979; Layzell et al., 1981) while it is 6.3–6.8 gC g N fixed<sup>-1</sup> for soybean, cowpea and white clover from nodulated roots at periods of intense N fixation (Ryle et al., 1979). The C cost determined by CO<sub>2</sub> released from nodulated roots is generally higher than that from nodules as the former includes root respiration. The C cost of N fixation also varies with growth stages (Ryle et al., 1979; Twary and Heichel, 1991), but it is a matter of debate that the C cost increases (Warembourg and Roumet, 1989; Voisin et al., 2003) or decreases (Adgo and Schulze, 2002) with the course of legume life cycle. In addition, the strain of Rhizobium may affect the C cost significantly. For example, the C cost in alfalfa (Medicago sativa L.) nodules formed by strain P207 is an average 9.4 molC molN $^{-1}$  which is 59% higher than that of nodules formed by strain I02F51, 5.9 molC molN $^{-1}$ (Twary and Heichel, 1991).

# 2.5 Seasonal Regulation of Legume Biological Nitrogen Fixation

The rate of legume BNF changes with physiological growth stages. It is low in the early growth stages while nodules are establishing (Lawrie and Wheeler, 1973) and reaches a maximum value between early flowering and early seed-filling, depending on the species and growing conditions (Lawn and Brun, 1974; Klucas, 1974; Nelson et al., 1984; Jensen, 1987). After

the peak, N fixation decreases dramatically or even ceases during seed-filling (Herridge and Pate, 1977; Beverly and Jarrell, 1984; Sinclair et al., 1987), due to nodule senescence (Lawrie and Wheeler, 1973) and poor C supply as a result of the strong demand for seed dry matter accumulation (Herridge and Pate, 1977; Voisin et al., 2003, 2007).

## 3 Quantification of Legume Biological Nitrogen Fixation

Legume BNF may be quantified by direct measurement, estimation based on yield or with empirical models, or simulation by crop models. The methods used to measure N fixation directly so far, such as the acetylene reduction/hydrogen increment assay, N difference, <sup>15</sup>N-labelling and ureide, have been thoroughly reviewed (Herridge et al., 2008; Carlsson and Huss-Danell, 2008). The review of these methods is beyond our focus, and in this paper we describe and compare only the methods to estimate and simulate N fixation.

#### 3.1 Estimation With Empirical Models

A static estimation of N fixation during the whole growing season may use either economic yield or above-ground dry matter. The equation is:

$$N_{fix} = \alpha \cdot DM \cdot f_{leg} \cdot N_{con} \cdot \% N df a \cdot (1 + R_{root})$$
(1)

where *DM* is the yield or aboveground dry matter,  $f_{leg}$ is the proportion of legume if it is intercropped,  $N_{con}$ is the N concentration in the legume, %Ndfa is the proportion of total plant N derived from N<sub>2</sub> fixation, and  $R_{root}$  is the ratio of the fixed N belowground to the fixed N aboveground. Values for the parameters  $f_{leg}$ ,  $N_{con}$  and  $R_{root}$  for different species at a range of sites are summarized in Table 2.  $\alpha$  is a parameter that has different definitions according to author. For example,  $\alpha$  may represent the decline in %Ndfa under high soil mineral N conditions, in terms of net mineral N input (g N m<sup>-2</sup>), and is calculated as:

$$\alpha = 1 - \beta \cdot N_{net.inorg} \tag{2}$$

where  $\beta$  evaluates the sensitivity of legume BNF to soil mineral N (Korsaeth and Eltun, 2000). This is set to 0.028 for white clover (*Trifolium repens* L.) and red clover (*Trifolium pratense* L.), 0.043 for grey peas (*Pisum arvense* L.) and common vetch (*Vicia sativa* L.). Alternatively,  $\alpha$  is used as a modifier to compensate for the proportion of fixed N transferred to a companion crop, consumed by grazing animals, or lost by immobilization (Høgh-Jensen et al., 2004).

The second method to estimate N fixation is by empirical models based on the correlation of fixed N in the final yield against variables, such as harvested dry matter or the proportion of legume in mixed leys. A linear equation has been fitted to the measured data of mixed white clover and grass swards at different sites from four countries. This showed a significant correlation between fixed N and the extra dry matter of mixed leys, compared with corresponding pure grass either for cut swards or grazed swards (Watson and Goss, 1997). Kristensen et al. (1995) found fixed N at harvest increased linearly with clover dry matter content in mixed swards through statistically analyzing the experimental data from different sites with distinct soil types and irrigation schemes. The equation for the estimation of N fixation (kg N ha<sup>-1</sup>) is summarized as,

$$N_{fix} = c + d \cdot Leg \tag{3}$$

where *Leg* is the extra dry matter increment (kg ha<sup>-1</sup>) in white clover mixed leys compared with a pure grass ley in Watson's estimation, and clover dry matter content (%clover) in clover mixed leys for Kristensen's study (1995), respectively; and *c* and *d* are parameters the values of which are summarized in Table 3.

The first method described above is apparently a direct estimation of N fixation, and the parameter values can be easily measured on site or estimated from the literature. It does not strictly require an adequate dataset for multiple years to determine the parameters, so it is easy to use. However, when determining the parameter values, data from years of abnormal weather should be avoided, and the properties of the soil should be relatively stable year-on-year. As these equations are independent from environmental factors such as soil properties and weather conditions, they are only applicable and accurate for similar sites and average weather conditions. In addition, the parameter values need to be adjusted if the equations are used for different sites or legumes.

Species	Condition	N <sub>con</sub> (%)	%Ndfa (%)	R <sub>root</sub>	Reference
White clover	Stock camp	4.9	61	_	
White clover	Gentle slope	4.9	82	-	
Subterranean clover	Gentle slope	4.9	82	-	Ledgard et al. (1987)
Subterranean clover	Steep slope	4.9	82	-	
Slender birds-foot trefoil	Steep slope	4.9	82	-	
Lucerne	Cut 1-2 years	2.7	74	0.25	
Red clover	Cut 1-2 years	3.0	74	0.25	
White clover	Intercropping with grass, cut 1–2	4.3	95	0.25	Høgh Jansan et al. (2004)
White clover	years Intercropping with grass, grazed 1–2 years	4.3	80	0.25	nggi-jensen et al. (2004)
Read clover	Intercropping with grass, cut 1–2 years	3.0	95	0.25	
Red clover	Intercropping with grass, grazed 1–2 years	3.0	80	0.25	
White clover	Intercropping with grass, cut >2 years	4.3	95	0.25	
White clover	Intercropping with grass, grazed >2 years	4.3	75	0.25	
White clover	1st cut	3.94	92.3	For all cuts,	Korsaeth and Eltun (2000)
	2nd cut	3.49	92.3	0.27 (year 1 and 2), 0 (year 3)	
Red clover	1st cut	3.14	92.3	-	
	2nd cut	2.91	92.3		
Grey peas		2.80	65.4	0.045	
Common vetch		3.00	65.4	0.27	

Table 2 Parameter values used to estimate N fixation in equation (1)

**Table 3** Values of parameters, c and d, in linear empirical models expressed in equation (3)

Condition	С	d	Model reference
Cut swards	6.8	0.067	Watson and Goss (1997)
Grazed swards	-168.1	0.067	
1-2 cropping years	18	4.47	Kristensen et al. (1995)
3-5 cropping years	19	2.77	

In contrast, the second method is based on statistical correlation and assumes that N fixation has a strong linear relationship to the variables. It is more flexible to use and can be applied to one specific site or multiple sites with different soil types, depending on how the empirical relationship is developed and which sites the data were obtained from. This approach has a higher data requirement compared to the first method, and the data should be representative and adequate to guarantee the correlation and determine the parameter values. However, as with the first method, these approaches are restricted to specific sites because the equation is not able to represent the interaction between plant and environment mechanistically.

# 3.2 Mechanistic Simulation of Legume Biological Nitrogen Fixation as a Subroutine in Crop Models

The simulation of legume BNF in soybean developed by Duffy et al. (1975), probing into the biophysiochemical transformations of N in tile-drained soil, might be the earliest mechanistic model involving leguminous N fixation. The rate of N fixation by soybean in the model depends on the root growth rate, which is derived from the sigmoid relationship with the number of days after planting.

	Facto	ors					
Model	$f_T$	$f_W$	$f_N$	$f_C$	fgro	Simulated legume specie	Reference
Sincloin Model*		$\checkmark$				soybean	Sinclair (1986)
Sincial Wodel		$\checkmark$			$\checkmark$	soybean, cowpea, black gram	Sinclair et al. (1987)
EPIC		$\checkmark$	$\checkmark$		$\checkmark$	soybean	Sharpley and Williams (1990); Bouniols et al. (1991); Cabelguenne et al. (1999)
Hurley Pasture Model	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		white clover	Thornley (1998); Thornley and Cannell (2000); Thornley (2001)
			$\checkmark$	$\checkmark$		field pea	Eckersten et al. (2006)
Schwinning Model*			$\checkmark$			white clover	Schwinning and Parsons (1996); Schmid et al. (2001)
CROPGRO	$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$	soybean, peanut, drybean, velvet bean, faba bean, cowpea	Boote et al. (1998); Sau et al. (1999); Hartkamp et al. (2002); Boote et al. (2002, 2008)
SOILN	$\checkmark$	$\checkmark$	$\checkmark$			white clover	Wu and McGechan (1999)
APSIM		$\checkmark$	$\checkmark$		$\checkmark$	soybean, chickpea, peanut, mungbean, lucerne	Herridge et al. (2001); Robertson et al. (2002)
Soussana Model*			$\checkmark$			white clover	Soussana et al. (2002)
STICS	$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	field pea and other legumes	Brisson et al. (2009); Corre-Hellou et al. (2007, 2009)

**Table 4** Simulation models that include legume BNF, and the factors considered in each model.\* indicates model named here after the first author's name.  $f_T$ ,  $f_W$ ,  $f_N$ ,  $f_C$  and  $f_{gro}$  are the fac-

tor of soil temperature, soil/plant water, soil/plant nitrogen, plant carbon and plant growth stage, respectively, used in equations (4, 5)

In more recent simulation models of BNF in legumes, the most popular method to estimate the rate of legume BNF is a potential or maximum fixation rate modified by the influence of environmental factors. The potential fixation rate is estimated based on either a demand-uptake mechanism or on the dry matter of plant tissues, and is varied with plant growth stages. The environmental factors normally include soil temperature, soil or plant water content, soil mineral N or substrate N concentration in plant tissues and substrate C concentration in the plant. Other factors, such as soil pH, salinity and the supply of other nutrients, have not been included in models to date.

In this paper we review the most-used recent simulation models in which a legume N fixation function has been implemented (Table 4). As crop models may be used under different circumstances, the estimation of N fixation may have various versions within the same model. The general expression of the calculation in the majority of the reviewed models can be written as:

$$N_{fix} = N_{fixpot} f_T f_W f_N f_C f_{gro} \tag{4}$$

while in the EPIC and STICS models it is:

$$N_{fix} = N_{fixpot} f_T \min(f_W, f_N) f_{gro}$$
(5)

where,  $N_{fixpot}$  is the potential N fixation rate (g N fixed day<sup>-1</sup>),  $f_T$  is the influence function of soil temperature,  $f_W$  is a soil water deficit or flooding function,  $f_N$  is the function of soil mineral N or root substrate N concentration,  $f_C$  is the function of substrate C concentration in plant or root,  $f_{gro}$  is the influence factor of growth stage and min is the mathematical function to take a minimum value between  $f_W$  and  $f_N$ . There is an extra function,  $f_a$ , representing the limitation by anoxia in the STICS model.

#### 3.2.1 Potential N Fixation Rate

There are two definitions on a potential legume BNF rate used in the models based either on the difference between N demand and uptake by a legume, or on the N fixation capacity of legume nodules.

The EPIC and APSIM models use variations of the first definition to estimate potential N fixation rate. The EPIC model assumes that the total plant N demand is equal to the potential N fixation (Bouniols et al., 1991; Cabelguenne et al., 1999). APSIM defines critical N concentrations for plant tissues and uses these to estimate N demand by maintaining non-stressed N levels in plant tissues and supporting the N demand of new tissues. This N demand is met by either N uptake from soil and/or N fixation. The former has a higher priority because the process is less energy consuming than N fixation (Macduff et al., 1996); N fixation is only calculated if N uptake can not meet the plant N demand. Thus the potential N fixation is assumed to be the difference between plant N demand and N uptake (Herridge et al., 2001; Robertson et al., 2002).

The second definition is based on the strong relationship between N fixation and either nodule size/biomass (Weisz et al., 1985; Voisin et al., 2003) or root biomass (Voisin et al., 2007). As the biomass of both nodules and roots are difficult to measure in the field, some studies have used above-ground biomass to replace nodule/root biomass, based on the relationship between these two variables (Denison et al., 1985; Bell et al., 1994; Yu et al., 2002). The potential fixation rates used in the models are shown in Table 5.

#### 3.2.2 Impact of Soil Temperature

The impact of soil temperature on N fixation rate is assumed to follow a four-threshold-temperature function in most of the reviewed models:

$$f_T = \begin{cases} 0 & (T < T_{min} \text{ or } T > T_{max}) \\ \frac{T - T_{min}}{T_{optL} - T_{min}} & (T_{min} \leqslant T \leqslant T_{optL}) \\ 1 & (T_{optL} \leqslant T \leqslant T_{optH}) \\ \frac{T_{max} - T}{T_{max} - T_{optH}} & (T_{optH} < T < T_{max}) \end{cases}$$
(6)

where *T* is soil temperature (°C),  $T_{min}$  is the minimum temperature below which N fixation ceases,  $T_{optL}$ and  $T_{optH}$  define the optimal temperature range within

Table 5 Potential fixed N rate used in the models. Value in APSIM varies with growth stage

		Maximum specific			
Model	Species	N fixation rate	Unit	Based on	Reference
Sinclair Model	soybean cowpea black gram	$\begin{array}{c} 0.55 - 0.7 \times 10^{-3} \\ 0.7 - 0.8 \times 10^{-3} \\ 0.7 - 0.75 \times 10^{-3} \end{array}$	g N gshoot $DM^{-1} d^{-1}$ g N gshoot $DM^{-1} d^{-1}$ g N gshoot $DM^{-1} d^{-1}$	shoot biomass shoot biomass shoot biomass	Sinclair (1986); Sinclair et al. (1987)
APSIM	legumes	1.0-6.0×10 <sup>-3</sup>	g N gshoot DM <sup>-1</sup> d <sup>-1</sup>	shoot biomass	(APSIM source code: http://apsrunet.apsim. info/svn/development/ trunk/apsim/)
	soybean, chick pea, mungbean, peanut, lucerne	N demand-uptake	g N d <sup>-1</sup>	N demand after N uptake	Herridge et al. (2001); Robertson et al. (2002)
EPIC	soybean	total N demand	$g N d^{-1}$	total N demand	Bouniols et al. (1991); Cabelguenne et al. (1999)
Hurley Pasture Model	clover	0.175	g N gnodule $DM^{-1} d^{-1}$	nodule biomass	Brugge and Thornley (1984)
	whiteclover	$0.05 - 1.0 \times 10^{-3}$	g N groot $DM^{-1} d^{-1}$	root biomass	Thornley, 1998; Thornley and Cannell (2000); Thornley (2001)
	pea	0.012-0.027	g N groot $DM^{-1} d^{-1}$	root biomass	Eckersten et al. (2006)
CROPGRO	faba bean soybean	0.05 0.045	g N gnodule $DM^{-1} d^{-1}$ g N gnodule $DM^{-1} d^{-1}$	nodule biomass nodule biomass	Boote et al. (2002)
SOILN	white clover	0.1106	g N gnodule $DM^{-1} d^{-1}$	nodule biomass	Wu and McGechan (1999)
Soussana Model	white clover	0.04	g N groot $DM^{-1}d^{-1}$	root biomass	Soussana et al. (2002)
STICS	legumes	-	g N gshoot DM <sup>-1</sup>	aboveground growth rate	Brisson et al. (2009)
	field pea	0.028	g N gshoot DM <sup>-1</sup>	aboveground growth rate	Corre-Hellou et al. (2007); Corre-Hellou et al. (2009)

Model	Species	$T_{min}$	$T_{optL}$	$T_{optH}$	$T_{max}$	Reference
CROPGRO	soybean	5	20	35	44	Boote et al. (2008)
	cowpea	5	20	35	44	
	velvet bean	5	23	35	44	
	dry bean	4	19	35	44	
	peanut	7	23	34	44	
	faba bean	1	16	25	40	
SOILN	white clover	9	13	26	30	Wu and McGechan (1999)
STICS	legumes	0	15	25	35	Brisson et al. (2009)

**Table 6** Values of threshold temperatures ( $^{\circ}$ C) used in three models that use equation (6) to simulate the effect of temperature on legume BNF

which the response function to soil temperature is the unit, and  $T_{max}$  is the maximum temperature above which fixation stops. The values of these four temperatures vary among models and are shown in Table 6.

A slightly different function is used in the Hurley Pasture model to simulate the influence of temperature on N fixation and most plant rate responses of white clover (Thornley, 1998):

$$f_T = T^2 (45 - T) \times 10^{-4}.$$
 (7)

There are variations in the response functions adopted in the reviewed models. For example, the functions for white clover in Hurley Pasture and SOILN differ in curve shapes and threshold temperatures (Fig. 1). There is a much wider range of temperature for N fixation in Hurley Pasture than SOILN, perhaps because the function in the Hurley Pasture model is not specially assigned to the N fixation module, but is used to simulate most plant processes. The function reaches unit at 20°C for both models and the maximum value



Fig. 1 The response of N fixation rate in white clover to temperature in the Hurley Pasture Model (solid line) and SOILN (dotted line)

of the function occurs at 30°C in Hurley Pasture, while the function is zero in SOILN at the same temperature. The value of the function in Hurley Pasture is higher than SOILN when the soil temperature is below 10°C, but this reverses when soil temperature is in the range 10 to 20°C. Over 20°C there is a large difference between Hurley Pasture and SOILN. This may be because Hurley Pasture was developed in southern Britain, where the temperature is often between 4 and 16°C (Thornley and Cannell, 2000) and thus it might not perform well at much higher temperatures.

#### 3.2.3 Impact of Soil Water Status

Soil water stress in the form of either a deficit or excess in the root zone could inhibit nodule nitrogenase activity. The effect of water deficit is considered in some models, but only few models take account for the influence of excess water. An exponential or linear equation derived from experimental data to represent the response of legume BNF rate to soil water deficit is incorporated into most of the reviewed models.

In STICS, the soil is divided into depth layers, and the proportion of these soil layers where water contents are above permanent wilting point is used as the water deficit factor.

In Sinclair's model, an empirical sigmoid relationship between relative nodule nitrogenase activity, expressed as  $C_2H_2$ , and transpirable soil water was developed based on experimental data (Sinclair, 1986; Sinclair et al., 1987):

$$f_W = -1 + \frac{2}{1 + e^{(-m * f_{TSW} + n)}} \tag{8}$$

where  $f_{TSW}$  is the fraction of transpirable soil water, and the total transpirable soil water content is defined as the difference between field capacity and the soil water content when the transpiration rate from drought-stressed plants decreases to less than 10% of that from well-watered plants (Sinclair, 1986); *m* and *n* are constants which defines the sensitivity of legume BNF to low soil water content. Values of *m* and *n* are 6 and 0 for soybean, 9 and 0.03 for both cowpea and black gram, respectively.

A linear function is incorporated into APSIM, EPIC (Sharpley and Williams, 1990; Bouniols et al., 1991; Cabelguenne et al., 1999) and SOILN (Wu and McGechan, 1999):

$$f_{w} = \begin{cases} 0 & (W_{f} \leq W_{a}) \\ \varphi_{1} + \varphi_{2} \cdot W_{f} & (W_{a} < W_{f} < W_{b}) \\ 1 & (W_{f} \geq W_{b}) \end{cases}$$
(9)

where  $W_f$  is the relative available soil water, expressed as the ratio of available soil water content to that at field capacity,  $\phi_1$  and  $\phi_2$  are coefficients, and  $W_b$  is a critical value of  $W_f$  above which N fixation is not inhibited by soil water content and  $W_a$  is the threshold below which N fixation is totally restricted by soil water deficit. The values for the parameters are shown in Table 7.

Although a linear expression similar to equation (9) is used in CROPGRO to simulate the impact of water stress on N fixation, the variable  $W_f$  is defined as the ratio of root water uptake to transpiration demand (Boote et al., 2008). Furthermore, the value of  $W_f$  on the current day is kept to compare with the average of  $W_f$  for the last eight days, to account for the prolonged impact of any previous severe drought on N fixation. The final value of  $W_f$  is equal to the lesser of these two values.

The upper soil layer above 30 cm depth is prone to be dry or wet when drought or irrigation occurs, and this causes a lag between water potential in the root nodules and that in the soil (Sprent, 1972; Albrecht et al., 1984). Therefore, water potential within roots could be a more stable indicator to quantify the relationship between water stress and N fixation. In the Hurley Pasture model, it is assumed that N fixation is controlled through chemical activity in roots, which is further influenced by root water potential and soil temperature. Thus the effect of water on N fixation is combined together with temperature (Thornley, 1998):

$$f_W = e^{20*\left[\frac{18*\varphi_{rt}}{8^{314*(T_s+273.15)}}\right]}$$
(10)

where  $\phi_{rt}$  is root water potential (J kg<sup>-1</sup>) and  $T_s$  is soil water temperature (°C).

Compared with water deficit, the influence of excessive water on the simulation of N fixation has received very little attention in published studies. In Sinclair's model, a simple assumption is adopted that N fixation is set to zero once flooding happens (Sinclair, 1986; Sinclair et al., 1987). In STICS, the restraint of legume BNF by excessive water is considered as a stress from anoxia, which is calculated as the proportion of soil layers that are in anaerobic conditions in the nodulation zone (Brisson et al., 2009). The evaluation of N fixation inhibition by water excess is incorporated into CROPGRO (Boote et al., 2008) by computing the fraction of pore space filled with water. When pore space is fully filled with water, N fixation is inhibited, but the validation of this rule to date is insufficient.

#### 3.2.4 Effect of Soil Mineral N or Internal Substrate N

A natural logarithmic function to represent the effect of soil mineral N content on N fixation by white clover is incorporated in SOILN model:

$$f_N = \begin{cases} 1 - 0.0784 \ln N_s & (N_s \ge 1) \\ 1 & (N_s < 1) \end{cases}$$
(11)

where  $N_s$  is soil mineral N concentration (mg N m<sup>-3</sup>).

As N uptake and N fixation are the main N source for legumes, the influence of soil mineral N on legume BNF rate could be presented indirectly by its influence on N uptake from soil. In Schwinning's model, such a scheme is used:

$$f_N = \varepsilon \times (1.0 - f_{Nup}) = \varepsilon \times \left(1.0 - f_{\max} \frac{1}{1 + K_N / N_s}\right)$$
(12)

where  $\varepsilon$  is the efficiency of legume BNF and a value of 0.6 was used for white clover,  $f_{max}$  is a maximal

Table 7 Paramete	er values in	the respo	nse function to	o soil water defi	icit								
Model	APSIM											EPIC	SOILN
Parameter value	soybean	peanut	navy bean	mung bean	lupin	lucerne	faba bean	cow pea	chick pea	butterfly pea	field pea	soybean	White clover
<u>ф1</u>	-0.33	0	0	0	0	0	0	0	0	0	0	-0.82	0.216
φ2	1.67	2.5	2	2	2	1.25	2	2	2	2	2	1.82	0.789
Wb	0.8	0.4	0.5	0.5	0.5	0.8	0.5	0.5	0.5	0.5	0.5	1	1
Wa	0.2	0	0	0	0	0	0	0	0	0	0	0.45	Ι

defic
water
soil
5
function
response
the
.п
values
Parameter
e

fraction of N derived from soil mineral N uptake, which is assumed to be 0.85 for white clover to make sure that N fixation still retains even in high soil mineral N concentrations in the root zone;  $K_N$  is the nitrate content (g N m<sup>-2</sup>) at which an N uptake rate reaches half its maximal rate and  $N_S$  is the actual soil nitrate content (g N m<sup>-2</sup>).

Even though the schemes in SOILN and Schwinning's model are different and  $f_N$  in Schwinning's model is never greater than 0.6 due to the limitation by legume BNF efficiency, they describe a similar trend of N fixation response to soil mineral N by white clover. The values of  $f_N$  from these two functions are very close while soil mineral N is higher than 0.05 g N m<sup>-2</sup> (Fig. 2).

The impact of soil mineral N is assessed as a linear function when soil nitrate concentration ( $N_{sNitra}$ ) is between 10 and 30 g N m<sup>-3</sup> within 0.3 m top soil in EPIC (Sharpley and Williams, 1990; Cabelguenne et al., 1999):

$$f_N = \begin{cases} 1 & (N_{sNitra} \le 10) \\ 1.5 - 0.05 N_{sNitra} & (10 < N_{sNitra} < 30) . \end{cases}$$
(13)

A similar function is incorporated in STICS as in EPIC, with a different value of  $N_{sNitra}$ . Moreover, in STICS, nodulation progress is also inhibited by high soil nitrate concentration, which is represented by a reduction of potential N fixation rate. Once the soil nitrate concentration in the nodulation zone is greater



**Fig. 2** N fixation response to soil mineral N concentration by white clover in the Schwinning Model (closed circle) and SOILN (open circle), assuming the nodulation zone is 40 cm deep

than a critical value,  $N_{fixpot}$  equals a baseline value; otherwise,  $N_{fixpot}$  is kept at the normal value (Brisson et al., 2009).

Both the Hurley Pasture Model and Soussana's model use plant substrate N concentration to simulate the influence of N nutritional status on N fixation rate:

$$f_N = \frac{1}{1 + N_{inter}/K_r} \tag{14}$$

where  $N_{inter}$  (g N g<sup>-1</sup>r.wt) is root substrate N concentration in the Hurley Pasture Model or plant substrate N concentration in Soussanna's model, respectively; and  $K_r$  is a coefficient inhibiting N fixation at high internal N concentration and it is set to 0.02 for white clover and 0.01 for field pea in the Hurley Pasture Model (Thornley, 1998, 2001; Eckersten et al., 2006), 0.004 for both a normal cultivar and a low-soil-N-uptake breeding cultivar, and 0.0004 for a low-N-fixation breeding cultivar in Soussana's model for white clover (Soussana et al., 2002).

#### 3.2.5 Influence of Plant Substrate C or C Supply

Biological N fixation has a high energy demand and the rate of legume BNF depends on the C supply, which is the energy source for N fixation. Experimental observations demonstrate that the enhancement of photosynthetic C allocated to roots which is available for N fixation and nodule establishment increases nodule numbers and N fixation rate (Voisin et al., 2003; Haase et al., 2007; Kirizii et al., 2007). However, among the reviewed models only the Hurley Pasture and CROP-GRO models implemented this mechanism.

In the Hurley Pasture Model, a Michaelis-Menten function is used to demonstrate the effect of root substrate C on the N fixation of white clover and field pea:

$$f_c = \frac{1}{1 + K_C / C_r}$$
(15)

where  $C_r$  is legume root C substrate concentration (gC g<sup>-1</sup> r.wt), and  $K_c$  is a Michaelis-Menten constant, set to be 0.01 for white clover (Thornley, 2001) and 0.035 for field peas (Eckersten et al., 2006).

In CROPGRO, photosynthetic carbohydrate supports not only the energy consumption of N fixation but is also the substrate supply for nodule biomass ac-



**Fig. 3** The representation of daily carbohydrate routes in nodules used in CROPGRO (Boote et al., 2008).  $[CH_2O]_n$  in dotted line means that the amount of  $[CH_2O]_n$  here is only used for calculation, neither for fixation nor nodule growth

cumulation. Carbohydrate allocated to nodulated roots is divided into three parts with priorities: requirement for minimum nodule growth, the cost for N fixation and the requirement for nodule normal growth (Fig. 3). First, it needs to guarantee the minimum daily nodule growth (step 1). Then the carbohydrate left over after the reduction for the minimum nodule growth is assumed to be potentially available for N fixation to determine potential N fixation rate (step 2), and in this step carbohydrate is not truly used to fix N until the actual N fixation rate is determined (step 3). If there is any carbohydrate left after the third step, the remainder will be used to produce more nodules (step 4) (Boote et al., 2008).

#### 3.2.6 Changes in N Fixation with Plant Growth Stage

The quantitative effect of plant growth stage on legume BNF rate is considered in very few models and in general the process is stopped forcibly after the legume attains a certain growth stage. For example, N fixation ceases at the beginning of seed growth for cowpea and black gram, but continues until the end of seed-filling for soybean in Sinclair's model (Sinclair et al., 1987), whereas it stops at physiological maturity in CROP-GRO (Boote et al., 2008).

A more specific function, similar to the temperature response function, is incorporated into EPIC and STICS to simulate the seasonal change of N fixation (Sharpley and Williams, 1990; Bouniols et al., 1991; Cabelguenne et al., 1999):

$$f_{gro} = \begin{cases} 0 & (g < g_{min} \text{ or } g > g_{max}) \\ \frac{g - g_{min}}{g_{optL} - g_{min}} & (g_{min} \leqslant g \leqslant g_{optL}) \\ 1 & (g_{optL} \leqslant g \leqslant g_{optH}) \\ \frac{g_{max} - g}{g_{max} - g_{optH}} & (g_{optH} < g < g_{max}) \end{cases}$$
(16)

where  $g_{min}$  is a period before which no fixation happens because of inadequate nodulation, expressed as the percentage of total growing period required;  $g_{optL}$  and  $g_{optH}$  are the beginning and end time within which legume BNF rate is not limited by growth stage; and  $g_{max}$  is the time after which N fixation ceases because of nodule senescence. The values of the parameters for soybean are set to 10%, 30%, 60%, and 80% of the life cycle by Bouniols et al. (1991), and 15%, 30%, 55% and 75% by Cabelguenne et al. (1999). In the STICS model,  $g_{min}$  and  $g_{max}$  are the times of nodule initiation and senescence respectively, and  $g_{optL}$  and  $g_{optH}$  correspond to 27% and 80% of nodule life (Brisson et al., 2009).

#### 3.3 Summary of Approaches

In addition to direct field measurements, estimates of legume BNF can be based on harvested yield or be derived from simple empirical models. However, these approaches generate data that is strictly valid only for the prevailing conditions at the time and place of measurement, with limited potential for use in wider-scale predictions. Simulation of legume BNF by models that incorporate the kinetics of N fixation may be the best approach as they may help us to understand the nature of the detailed relationships between N fixation and environmental and plant factors. Thus they permit more accurate simulation of legume BNF under a much broader range of farming systems and environmental conditions than other approaches.

Our review has found that the method used to simulate legume BNF most frequently in recent publications is to modify potential N fixation rate by factors such as soil temperature, soil/plant water, soil/plant N, plant C and plant growth stage. However, which of these factors are included and the functions used to relate them to the potential N fixation rate differ between models. Despite this range of approaches, the simulation of legume BNF in recent models may be conveniently summarized as:

(i) Potential N fixation rate may be estimated based on plant N demand and N uptake from the soil, in which case a critical plant N concentration needs to be defined appropriately for each species. Alternatively, if the potential N fixation rate is estimated from nodule/root/aboveground dry matter, then the maximum specific N fixation rate has to be estimated experimentally, as this value varies widely between species and sites.

- (ii) The influence of soil temperature on legume BNF is commonly represented by a four-thresholdtemperature function, or a sigmoidal cubic function in the Hurley Pasture Model (Thornley, 1998). The values of these four thresholdtemperatures vary with species and cultivar.
- (iii) The relationship between legume BNF rate and soil water deficit, alternatively represented by low plant water status in some models, is described by either exponential or linear equations. The choice of equations has only a small effect on the output, unless some species are more sensitive than others to water deficit and this is likely as species will have been bred for different environment. However, the effect of excessive water on N fixation is over-simplified or absent in all models and this factor needs more attention.
- (iv) The sensitivity of legume BNF to a factor representing general N availability, e.g. soil mineral N or plant N content, is the function that differs most between different models. In the EPIC model (Sharpley and Williams, 1990; Cabelguenne et al., 1999), legume BNF is unaffected until soil mineral N in the root zone is higher than 10 g N m<sup>-3</sup> whereas in SOILN (Wu and McGechan, 1999) and Schwinning's model (1996) legume BNF is highly sensitive to even small amounts of soil mineral N. This latter pattern is also seen in the Hurley Pasture Model (Thornley, 2001) and Soussana's model (2002), which use root and plant N concentrations respectively instead of soil mineral N concentration.
- (v) Factors that increase plant C status have a positive effect on legume BNF. The relationships are sometimes represented by Michaelis-Menten functions, or are included in more complex simulations together with nodule growth and the synthesis of the products of legume BNF.
- (vi) The change in legume BNF with legume growth stage follows a function that contains four threshold values in most of the models. However, in AP-SIM, the same effect is achieved by using different values of potential specific N fixation rate for each growth stage.

#### 4 Discussion

There has been much progress in simulating N fixation by forage and grain legumes since the first simple model by Duffy et al. (1975). However, the analysis presented above has revealed areas of the simulation schemes that are in need of refinement, and identified some potentially important factors omitted from all current models. These areas, and suggestions for their improvement, are discussed below.

It is preferable to define the key parameter of potential N fixation rate based on nodule mass rather than on the plant N deficit after soil N uptake. This is because normally legume BNF is not completely inhibited by soil mineral N and some species and cultivars continue to fix N even when soil mineral N is relatively high (Lamb et al., 1995; Blumenthal and Russelle, 1996). However, not all the models have the potential to estimate nodule biomass accurately. In these cases root or shoot biomass is used instead of nodule biomass to estimate potential N fixation rate, and this is based on the strong relationships between nodule size and plant size (Denison et al., 1985; Sinclair, 1986), and between nodule biomass and root biomass (Macduff et al., 1996). On this basis a constant ratio of nodule biomass to root or shoot biomass is assumed, and linear functions are used to calculate the potential N fixation rates in these models. In practice, however, this ratio varies with legume growth stage (Schiltz et al., 2005; Bollman and Vessey, 2006) and is affected by other environmental factors, such as soil mineral N (Voisin et al., 2002; Bollman and Vessey, 2006). Thus the constant ratio of nodule biomass to root or shoot biomass assumed in some models does not represent the real situation and would be better defined as a temporal factor that changes with environmental conditions. The importance of this is shown by the fact that even when the nodule biomass is available and is used to estimate potential legume BNF, the relationship between legume BNF rate and nodule mass varies with growth stage (Voisin et al., 2003).

The above section has shown how measurements of plant or nodule biomass may be used to modify the maximum potential legume BNF to provide an estimate of legume BNF in the field. However, a potentially more accurate approach is to use the amount and allocation pattern of photosynthetically-fixed C in the plant. Carbon is required for nodule growth, as an energy source for legume BNF and for the synthesis of N compounds produced from legume BNF (Minchin and Pate, 1973; Voisin et al., 2003), and many studies have shown a strong correlation between legume BNF and the C respired by nodules, nodulated roots (Mahon, 1977a, b; Ryle et al., 1979), or photosynthetic allocation (Lawrie and Wheeler, 1973; Voisin et al., 2003, 2007). Of all the models considered here, C sink is closely related to nodule growth and legume BNF only in CROPGRO (Boote et al., 2008). Thus improved experimental evidence of C allocation and usage by nodules and incorporation of these processes more widely into models would do much to improve estimates of potential legume BNF.

As mentioned above, soil mineral N in the rhizosphere inhibits legume root nodulation and legume BNF, and this inhibition is more severe under higher N concentrations, although moderate levels of soil mineral N stimulate nodulation and N fixation in some circumstances. It is difficult to give the critical value of soil N concentration below which nodulation and N fixation are stimulated (Munns, 1970; Streeter, 1988; Waterer et al., 1992; Waterer and Vessey, 1993b), and the issue about "starter N" is still controversial (Bethlenfalvay and Phillips, 1978; Bethlenfalvay and Phillips, 1978; Waterer and Vessey, 1993b; Gulden and Vessey, 1997). Taken together, these factors explain why this stimulating effect of soil mineral N is not incorporated into any of the current models of legume BNF, and suggest that much further experimental work may be needed before it is possible to do so. The lack of inclusion of this process is a probable cause of the large discrepancy between modelled and actual legume BNF in some circumstances.

A further process that is absent from the current models of legume BNF is that they cannot distinguish the different inhibitory effects of soil nitrate and ammonium in the rhizosphere. Experimentally it has been shown that legume BNF is less sensitive to ammonium than nitrate (Svenning et al., 1996; Bollman and Vessey, 2006; Gan et al., 2004), but all models use soil nitrate or total mineral N concentration.

Our analysis of the currently-used models of legume BNF has revealed that the inhibition of legume BNF by soil mineral N concentration might be overestimated in the model that uses nodule biomass to determine the potential N fixation rate,  $N_{fixpot}$ . Considering equation (4),  $N_{fixpot}$  is the potential N fixation rate, calculated as the amount of fixed N per unit of nodule biomass multiplied by nodule biomass. Thus

if nodule biomass is reduced by high soil mineral N concentration,  $N_{fixpot}$  will also be reduced. However, this model also uses  $f_N$ , the factor that relates legume BNF to soil mineral N concentration. This factor is derived experimentally from measurements of legume BNF under different levels of soil mineral N concentration, and by default this includes any inhibitory effect of soil mineral N concentration on legume BNF. Thus in STICS (Brisson et al., 2009), the effect of soil mineral N concentration on nodule biomass is effectively incorporated twice. We suggest that to avoid this problem, future models could remove the effect of nodule biomass on  $f_N$  by basing estimates of the latter on measurements taken from plants with well developed nodules. The effect of high soil mineral N concentration on nodule biomass should be included as it is currently, by its effect on  $N_{fixpot}$ .

In both experimental work and models of legume BNF, the effect of the water status of either soil or plant on legume BNF has concentrated on cases of water deficit rather than flooding. Despite many studies on the effect of soil water deficit on legume BNF, the nature of the relationship between N fixation and water soil status is poorly characterised. A linear or exponential function has been widely used in modelling studies to date except in the Hurley Pasture Model. CROP-GRO includes an eight-day moving average of a water factor, which is used to simulate the prolonged effect of drought on N fixation. This is consistent with the conclusion that the inhibition of N fixation by drought in peanut and cowpea is not alleviated immediately once the drought releases (Venkateswarlu et al., 1990). Some studies have found that drought stress on N fixation is caused more by the inhibition of photosynthesis than by drought directly (Huang et al., 1975), which highlights the importance of the link between N fixation and photosynthesis. However, some studies have shown that N fixation is more sensitive than photosynthesis to drought (Sprent, 1976) and N fixation may decline before transpiration or photosynthesis decreases (Sinclair et al., 1987; Sall and Sinclair, 1991; Purcell et al., 1997).

Stress from excessive water is omitted or oversimplified in all models, even though the proportion of water in pore spaces has been suggested as an approach to evaluate the restriction on nodule nitrogenase activity by excessive water in CROPGRO. More studies are needed on the effect of excess water on legume BNF in the future, both experimentally and in simulations.

In summary, the difficulty of simulating legume BNF is because of the large variance in N fixation between sites and species, and over time. Biological N fixation is a highly complex process which integrates plant and soil processes in the macro-environment with the micro-environmental processes of rhizobial bacteria in nodules (Nelson and Child, 1981; Bolaños et al., 1994; Rice et al., 2000). It is difficult to represent these two completely different but closely related processes in one model. Also, there is currently inadequate information to quantify reliably the relationship of legume BNF with those factors such as soil water and plant C. This is because legume BNF is not only sensitive to temperature, water, N and C cycling, which have been incorporated into models, but also to soil pH (Rechcigl et al., 1987; Peoples et al., 1995; Ibekwe et al., 1997; Rice et al., 2000; Le Roux et al., 2008), O<sub>2</sub> permeability (Cowan, 1978; Weisz and Sinclair, 1987; Faurie and Soussana, 1993; Serraj et al., 2001), salinity (Serraj and Drevon, 1998) and other nutrition regimes (Lynd et al., 1984; Collins et al., 1986; Bolaños et al., 1994; Chaudhary et al., 2008), which are currently absent from models despite frequently being cited as important factors.

Furthermore, in terms of the practical uses and benefits of N fixed by legumes, realistic field scenarios should be considered. In temperate agricultural systems, mixed pastures of legumes and grasses are common; intercrops of forage and grain legumes with cereal crops for animal feed are increasing in importance, especially in low-input and organic farming systems (Hauggaard-Nielsen et al., 2001, 2003; Andersen et al., 2004; Corre-Hellou et al., 2006). In these systems the interactions, both competitive and facilitative, between the components alter legume BNF in mixtures compared to monocultures (Malézieux et al., 2009). Intercropping enhances the proportion of legume N derived from fixation due to the higher competition for soil N. However, the total legume BNF is often reduced as a consequence of its dependence on not only soil N but also legume growth. In addition to deliberate intercropping, weeds have a similar influence on legume BNF. This may have a big influence on yield and N accumulation of the main crops in pure legume stands if the weeds are relatively competitive (Hauggaard-Nielsen et al., 2001, 2006). Schwinning and Parsons (1996) and Thornley (2001) incorporate the interaction between legumes and grasses in their models, which have been reviewed by Malézieux's (2009). Therefore,

these simulation schemes of interspecies competition are not reviewed here as we have focused on legumes only.

Apart from intercropping, other field managements such as grazing by livestock, especially in forage legume grasslands, affects N fixation through its effects on both legume growth and soil mineral N cycling due to urine and dung inputs. Thornley and Cannell (2000) simulated mineral N availability in grazed grassland with legume BNF through the effect on leaf area by grazing in the Hurley Pasture Model. Schwinning and Parsons (1996) implement not only the effect of grazing on leaf cover but also the urine deposits from animals into their model. However, their simulation scheme of N fixation is simpler than in other models as it considers only the direct influence of soil mineral N on BNF. Many of these dynamic and competitive aspects with grazing and other crops should be better incorporated into future models that aim to have real agricultural relevance.

#### 5 Conclusion

This review critically interprets the methods used to quantify legume BNF by the most commonly used experimental and modelling approaches. In particular, it highlights and compares the functions used to simulate legume BNF by different models and assesses their relative strengths in predicting N fixation with varying biotic and abiotic factors. We found that:

- (i) As a base to estimate legume BNF, all models use the potential N fixation rate although they differ in how they estimate this. Different functions are incorporated into each of the models to estimate how the maximum potential N fixation rate is restrained by soil temperature, soil water, soil/plant N concentration, plant C status and legume growth stages.
- (ii) Estimations of potential N fixation rate based on above ground biomass or root biomass are not as reliable as those based on nodule biomass (although see (iii) below). C supply from photosynthesis is recommended as the factor best able to explain the potential N fixation rate.
- (iii) The stimulating effect on legume BNF at relatively low levels of soil mineral N should be

distinguished from the inhibition of legume BNF by soil mineral N. The simulated inhibition of legume BNF by soil mineral N might be exaggerated if potential N fixation is modified by functions that include nodule biomass and the effect of soil mineral N on potential N fixation ( $f_N$ ).

- (iv) More experimental work is needed to characterise the effect of both soil water deficit and excess soil water on legume BNF. The responses of legume BNF to other factors currently absent from all models, such as soil pH and O<sub>2</sub> permeability, need to be included and reinforced with adequate experimental work.
- (v) Intercropping of legumes, either with grain crops or in grasslands, as well as the presence of grazing livestock, affect legume BNF in the fields. Models of legume BNF should take better account of these important practical uses of legumes.

Acknowledgements Y. Liu thanks the China Scholarship Council for financial support. SAC receives funding from the Rural and Environment Research and Analysis Directorate (RERAD) of the Scottish Government. The authors thank Drs M. Robertson and P. Devoil for the source code of the APSIM model, and N. Brisson and M. Launay for details of the STICS model.

#### References

- Abdel Wahab A.M., Zahran H.H., Abd-Alla M.H. (1996) Roothair infection and nodulation of four grain legumes as affected by the form and the application time of nitrogen fertilizer, Folia Microbiol. 41, 303–308.
- Adgo E., Schulze J. (2002) Nitrogen fixation and assimilation efficiency in Ethiopian and German pea varieties, Plant Soil 239, 291–299.
- Albrecht S.L., Bennett J.M., Boote K.J. (1984) Relationship of nitrogenase activity to plant water stress in field-grown soybeans, Field Crops Res. 8, 61–71.
- Andersen M.K., Hauggaard-Nielsen H., Ambus P., Jensen E.S. (2004) Biomass production, symbiotic nitrogen fixation and inorganic N use in dual and tri-component annual intercrops, Plant Soil 266, 273–287.
- Bell M.J., Wright G.C., Suryantini, Peoples M.B. (1994) The N<sub>2</sub>-fixing capacity of peanut cultivars with differing assimilate partitioning characteristics, Aust. J. Agric. Res. 45, 1455–1468.
- Bethlenfalvay G.J., Phillips D.A. (1978) Interactions between symbiotic nitrogen fixation, combined-N application, and photosynthesis in *Pisum sativum*, Physiol. Plant. 42, 119–123.
- Bethlenfalvay G.J., Abu-Shakra S.S., Phillips D.A. (1978) Interdependence of nitrogen nutrition and photosynthesis in *Pisum sativum* L. I. effect of combined nitrogen on symbiotic

nitrogen fixation and photosynthesis, Plant Physiol. 62, 127-130.

- Beverly R.B., Jarrell W.M. (1984) Cowpea response to N form, rate, and timing of application, Agron. J. 76, 663–668.
- Blumenthal J.M., Russelle M.P. (1996) Subsoil nitrate uptake and symbiotic dinitrogen fixation by alfalfa, Agron. J. 88, 909–915.
- Bolaños L., Esteban E., de Lorenzo C., Fernández-Pascual M., de Felipe M.R., Gárate A., Bonilla I. (1994) Essentiality of boron for symbiotic dinitrogen fixation in pea (*Pisum-sativum*) rhizobium nodules, Plant Physiol. 104, 85–90.
- Bollman M.I., Vessey J.K. (2006) Differential effects of nitrate and ammonium supply on nodule initiation, development, and distribution on roots of pea (*Pisum sativum*), Can. J. Bot. 84, 893–903.
- Boote K.J., Hoogenboom G., Jones J.W., Ingram K.T. (2008) Modeling nitrogen fixation and its relationship to nitrogen uptake in the CROPGRO model, in: Ma L., Ahuja L.R., Bruulsema T.W. (Eds.), Quantifying and Understanding Plant Nitrogen Uptake for Systems Modeling. CRC Press, Florence, USA, pp. 13–46.
- Boote K.J., Jones J.W., Hoogenboom G., Pickering N.B. (1998) The CROPGRO model for grain legumes, in: Tsuji G.Y., Hoogenboom G., Thornton P.K. (Eds.), Understanding Options for Agricultural Production, pp. 99–128.
- Boote K.J., Mínguez M.I., Sau F. (2002) Adapting the CROP-GRO legume model to simulate growth of faba bean, Agron. J. 94, 743–756.
- Bouniols A., Cabelguenne M., Jones C.A., Chalamet A., Charpenteau J.L., Marty J.R. (1991) Simulation of soybean nitrogen nutrition for a silty clay soil in southern France, Field Crops Res. 26, 19–34.
- Brisson N., Launay M., Mary B., Beaudoin N. (2009) Nitrogen transformations, in: Brisson N., Launay M., Mary B., Beaudoin N. (Eds.), Conceptual basis, formalisations and parameterization of the STICS crop model, Quæ, Versailles Cedex, France, pp. 141–165.
- Brugge R., Thornley J.H.M. (1984) Shoot root nodule partitioning in a vegetative legume – A model, Ann. Bot. 54, 653–671.
- Cabelguenne M., Debaeke P., Bouniols A. (1999) EPICphase, a version of the EPIC model simulating the effects of water and nitrogen stress on biomass and yield, taking account of developmental stages: validation on maize, sunflower, sorghum, soybean and winter wheat, Agric. Syst. 60, 175–196.
- Cannell M.G.R., Thornley J.H.M. (2000) Modelling the components of plant respiration: some guiding principles, Ann. Bot. 85, 45–54.
- Carlsson G., Huss-Danell K. (2008) How to quantify biological nitrogen fixation in forage legumes in the field, in: Dakora F.D., Chimphango S.B.M., Valentine A.J., Elmerich C., Newton W.E. (Eds.), Biological Nitrogen Fixation: Towards Poverty Alleviation through Sustainable Agriculture, Springer Netherlands, Dordrecht, pp. 47–48.
- Chalifour F.P., Nelson L.M. (1987) Effects of continuous combined nitrogen supply on symbiotic dinitrogen fixation of faba bean and pea inoculated with different rhizobial isolates, Can. J. Bot. 65, 2542–2548.
- Chaudhary M., Adu-Gyamfi J., Saneoka H., Nguyen N., Suwa R., Kanai S., El Shemy H., Lightfoot D., Fujita K. (2008) The effect of phosphorus deficiency on nutrient uptake, nitrogen

fixation and photosynthetic rate in mashbean, mungbean and soybean, Acta Physiol. Plant. 30, 537–544.

- Collins M., Lang D.J., Kelling K.A. (1986) Effects of phosphorus, potassium, and sulfur on alfalfa nitrogen-fixation under field conditions, Agron. J. 78, 959–963.
- Corre-Hellou G., Brisson N., Launay M., Fustec J., Crozat Y. (2007) Effect of root depth penetration on soil nitrogen competitive interactions and dry matter production in pea–barley intercrops given different soil nitrogen supplies, Field Crops Res. 103, 76–85.
- Corre-Hellou G., Faure M., Launay M., Brisson N., Crozat Y. (2009) Adaptation of the STICS intercrop model to simulate crop growth and N accumulation in pea-barley intercrops, Field Crops Res. 113, 72–81.
- Corre-Hellou G., Fustec J., Crozat Y. (2006) Interspecific competition for soil N and its interaction with N<sub>2</sub> fixation, leaf expansion and crop growth in pea–barley intercrops, Plant Soil 282, 195–208.
- Cowan M. (1978) The influence of nitrate, water potential and oxygen tension on nitrogen fixation in detached pea roots, Plant Foods Hum. Nutr. 28, 65–69.
- Dart P., Day J. (1971) Effects of incubation temperature and oxygen tension on nitrogenase activity of legume root nodules, Plant Soil 35, 167–184.
- Denison R.F., Weisz P.R., Sinclair T.R. (1985) Variability among plants in dinitrogen fixation (acetylene reduction) rates by field-grown soybeans, Agron. J. 77, 947–950.
- Duffy J., Chung C., Boast C., Franklin M. (1975) A simulation model of biophysiochemical transformations of nitrogen in tile-drained corn belt soil, J. Environ. Qual. 4, 477–486.
- Eaglesham A.R.J. (1989) Nitrate inhibition of root-nodule symbiosis in doubly rooted soybean plants, Crop Sci. 29, 115–119.
- Eckersten H., Af Geijersstam L., Torssell B. (2006) Modelling nitrogen fixation of pea (*Pisum sativum* L.), Acta Agric. Scand. B 56, 129–137.
- Engin M., Sprent J.I. (1973) Effects of water stress on growth and nitrogen-fixing activity of *Trifolium repens*, New Phytol. 72, 117–126.
- Evans J., McNeill A.M., Unkovich M.J., Fettell N.A., Heenan D.P. (2001) Net nitrogen balances for cool-season grain legume crops and contributions to wheat nitrogen uptake: a review, Aust. J. Exp. Agric. 41, 347–359.
- Evans J., O'Connor G.E., Turner G.L., Coventry D.R., Fettell N., Mahoney J., Armstrong E.L., Walsgott D.N. (1989)  $N_2$  fixation and its value to soil N increase in lupin, field pea and other legumes in south-eastern Australia, Aust. J. Agric. Res. 40, 791–805.
- Evans J., Turner G.L., O'Connor G.E., Bergersen F.J. (1987) Nitrogen fixation and accretion of soil nitrogen by fieldgrown lupins (*Lupinus angustifolius*), Field Crops Res. 16, 309–322.
- Faurie O., Soussana J.F. (1993) Oxygen-induced recovery from short-term nitrate inhibition of  $N_2$  fixation in white clover plants from spaced and dense stands, Physiol. Plant. 89, 467–475.
- Flechard C.R., Ambus P., Skiba U., Rees R.M., Hensen A., van Amstel A., van den Pol-van Dasselaar A., Soussana J.-F., Jones M., Clifton-Brown J., Raschi A., Horvath L., Neftel A., Jocher M., Ammann C., Leifeld J., Fuhrer J., Calanca P., Thalman E., Pilegaard K., Di Marco C., Campbell C.,

Nemitz E., Hargreaves K.J., Levy P.E., Ball B.C., Jones S.K., van de Bulk W.C.M., Groot T., Blom M., Domingues R., Kasper G., Allard V., Ceschia E., Cellier P., Laville P., Henault C., Bizouard F., Abdalla M., Williams M., Baronti S., Berretti F., Grosz B. (2007) Effects of climate and management intensity on nitrous oxide emissions in grassland systems across Europe, Agric. Ecosyst. Environ. 121, 135–152.

- Frame J., Newbould P. (1986) Agronomy of white clover, Adv. Agron. 40, 1–88.
- Gan Y., Stulen I., van Keulen H., Kuiper P.J.C. (2004) Low concentrations of nitrate and ammonium stimulate nodulation and N<sub>2</sub> fixation while inhibiting specific nodulation (nodule DW  $g^{-1}$  root dry weight) and specific N<sub>2</sub> fixation (N<sub>2</sub> fixed  $g^{-1}$  root dry weight) in soybean, Plant Soil 258, 281–292.
- Garg N., Geetanjali (2007) Symbiotic nitrogen fixation in legume nodules: process and signaling. A review, Agron. Sustain. Dev. 27, 59–68.
- Gibson A. (1971) Factors in the physical and biological environment affecting nodulation and nitrogen fixation by legumes, Plant Soil 35, 139–152.
- Goh K.M., Bruce G.E. (2005) Comparison of biomass production and biological nitrogen fixation of multi-species pastures (mixed herb leys) with perennial ryegrass-white clover pasture with and without irrigation in Canterbury, New Zealand, Agric. Ecosyst. Environ. 110, 230–240.
- Goh K.M., Edmeades D.C., Robinson B.W. (1978) Field measurements of symbiotic nitrogen fixation in an established pasture using acetylene reduction and a <sup>15</sup>N method, Soil Biol. Biochem. 10, 13–20.
- Gulden R., Vessey J.K. (1997) The stimulating effect of ammonium on nodulation in *Pisum sativum* L. is not long lived once ammonium supply is discontinued, Plant Soil 195, 195–205.
- Haase S., Neumann G., Kania A., Kuzyakov Y., Römheld V., Kandeler E. (2007) Elevation of atmospheric CO<sub>2</sub> and N-nutritional status modify nodulation, nodule-carbon supply, and root exudation of *Phaseolus vulgaris* L., Soil Biol. Biochem. 39, 2208–2221.
- Halliday J. (1975) An interpretation of seasonal and short term fluctuations in nitrogen fixation, PhD thesis, University of Western Australia, Perth.
- Halliday J., Pate J.S. (1976) The acetylene reduction assay as a means of studying nitrogen fixation in white clover under sward and laboratory conditions, Grass Forage Sci. 31, 29–35.
- Hartkamp A.D., Hoogenboom G., White J.W. (2002) Adaptation of the CROPGRO growth model to velvet bean (*Mucuna pruriens*): I. Model development, Field Crops Res. 78, 9–25.
- Hauggaard-Nielsen H., Ambus P., Jensen E.S. (2001) Interspecific competition, N use and interference with weeds in peabarley intercropping, Field Crops Res. 70, 101–109.
- Hauggaard-Nielsen H., Ambus P., Jensen E.S. (2003) The comparison of nitrogen use and leaching in sole cropped versus intercropped pea and barley, Nutr. Cycl. Agroecosyst. 65, 289–300.
- Hauggaard-Nielsen H., Andersen M.K., Jørnsgaard B., Jensen E.S. (2006) Density and relative frequency effects on competitive interactions and resource use in pea–barley intercrops, Field Crops Res. 95, 256–267.

- Havelka U.D., Boyle M.G., Hardy R.W.F. (1982) Biological nitrogen fixation, in: Stevenson F.J. (Ed.), Nitrogen in Agricultural Soils, ASA, Madison, Wisconsin, USA, pp. 365–422.
- Herridge D.F., Pate J.S. (1977) Utilization of net photosynthate for nitrogen fixation and protein production in an annual legume, Plant Physiol. 60, 759–764.
- Herridge D.F., Roughley R.J., Brockwell J. (1984) Effect of rhizobia and soil nitrate on the establishment and functioning of the soybean symbiosis in the field, Aust. J. Agric. Res. 35, 149–161.
- Herridge D.F., Turpin J.E., Robertson M.J. (2001) Improving nitrogen fixation of crop legumes through breeding and agronomic management: analysis with simulation modelling, Aust. J. Exp. Agric. 41, 391–401.
- Herridge D.F., Peoples M.B., Boddey R.M. (2008) Global inputs of biological nitrogen fixation in agricultural systems, Plant Soil 311, 1–18.
- Høgh-Jensen H., Loges R., Jørgensen F.V., Vinther F.P., Jensen E.S. (2004) An empirical model for quantification of symbiotic nitrogen fixation in grass-clover mixtures, Agric. Syst. 82, 181–194.
- Hossain S.A., Strong W.M., Waring S.A., Dalal R.C., Weston E.J. (1996) Comparison of legume-based cropping systems at Warra, Queensland. II. Mineral nitrogen accumulation and availability to the subsequent wheat crop, Aust. J. Soil Res. 34, 289–297.
- Howard J.B., Rees D.C. (1996) Structural basis of biological nitrogen fixation, Chem. Rev. 96, 2965–2982.
- Huang C.Y., Boyer J.S., Vanderhoef L.N. (1975) Limitation of acetylene reduction (nitrogen fixation) by photosynthesis in soybean having low water potentials, Plant Physiol. 56, 228–232.
- Ibekwe A.M., Angle J.S., Chaney R.L., van Berkum P. (1997) Enumeration and N<sub>2</sub> fixation potential of *Rhizobium leguminosarum* biovar *trifolii* grown in soil with varying pH values and heavy metal concentrations, Agric. Ecosyst. Environ. 61, 103–111.
- Imsande J. (1986) Inhibition of nodule development in soybean by nitrate or reduced nitrogen, J. Exp. Bot. 37, 348–355.
- Jensen E.S. (1986) The influence of rate and time of nitrate supply on nitrogen fixation and yield in pea (*Pisum sativum* L.), Fert. Res. 10, 193–202.
- Jensen E.S. (1987) Seasonal patterns of growth and nitrogen fixation in field-grown pea, Plant Soil 101, 29–37.
- Jensen E.S. (1997) The role of grain legume N<sub>2</sub> fixation in the nitrogen cycling of temperate cropping systems, DSc Thesis, Riso-R-885, Risø National Laboratory, Denmark.
- King B.J., Layzell D.B., Canvin D.T. (1986) The role of dark carbon dioxide fixation in root nodules of soybean, Plant Physiol. 81, 200–205.
- Kirizii D.A., Vorobei N.A., Kots' S.Ya. (2007) Relationships between nitrogen fixation and photosynthesis as the main components of the productivity in alfalfa, Russ. J. Plant. Physiol. 54, 589–594.
- Kishinevsky B.D., Sen D., Weaver R.W. (1992) Effect of high root temperature on *Bradyrhizobium*-peanut symbiosis, Plant Soil 143, 275–282.
- Klucas R.V. (1974) Studies on soybean nodule senescence, Plant Physiol. 54, 612–616.
- Korsaeth A., Eltun R. (2000) Nitrogen mass balances in conventional, integrated and ecological cropping systems and the re-

lationship between balance calculations and nitrogen runoff in an 8-year field experiment in Norway, Agric. Ecosyst. Environ. 79, 199–214.

- Kristensen E.S., Høgh-Jensen H., Kristensen I.S. (1995) A simple model for estimation of atmospherically-derived nitrogen in grass-clover systems, Biol. Agric. Hortic. 12, 263–276.
- Kuo T., Boersma L. (1971) Soil water suction and root temperature effects on nitrogen fixation in soybeans, Agron. J. 63, 901–904.
- Lamb J.F.S., Barnes D.K., Russelle M.P., Vance C.P., Heichel G.H., Henjum K.I. (1995) Ineffectively and effectively nodulated alfalfas demonstrate biology nitrogen fixation continues with high nitrogen fertilization, Crop Sci. 35, 153–157.
- Lawn R.J., Brun W.A. (1974) Symbiotic nitrogen fixation in soybeans. I. effect of photosynthetic source-sink manipulations, Crop Sci. 14, 11–16.
- Lawrie A.C., Wheeler C.T. (1973) The supply of photosynthetic assimilates to nodules of *Pisum Sativum L*. in relation to the fixation of nitrogen, New Phytol. 72, 1341–1348.
- Layzell D.B., Pate J.S., Atkins C.A., Canvin D.T. (1981) Partitioning of carbon and nitrogen and the nutrition of root and shoot apex in a nodulated legume, Plant Physiol. 67, 30–36.
- Layzell D.B., Rainbird R.M., Atkins C.A., Pate J.S. (1979) Economy of photosynthate use in nitrogen-fixing legume nodules: observations on two contrasting symbioses, Plant Physiol. 64, 888–891.
- Le Roux M.R., Khan S., Valentine A.J. (2008) Organic acid accumulation may inhibit N<sub>2</sub> fixation in phosphorus-stressed lupin nodules, New Phytol. 177, 956–964.
- Ledgard S.F., Steele K.W. (1992) Biological nitrogen fixation in mixed legume/grass pastures, Plant Soil 141, 137–153.
- Ledgard S.F., Brier G.J., Littler R.A. (1987) Legume production and nitrogen fixation in hill pasture communities, New Zeal. J. Agric. Res. 30, 413–421.
- Lie T.A. (1971) Symbiotic nitrogen fixation under stress conditions, Plant Soil 35, 117–127.
- Lindemann W.C., Ham G.E. (1979) Soybean plant growth, nodulation, and nitrogen fixation as affected by root temperature, Soil Sci. Soc. Am. J. 43, 1134–1137.
- Lynd J.Q., Hanlon E.A., Jr., Odell G.V., Jr. (1984) Nodulation and nitrogen fixation by arrowleaf clover: effects of phosphorus and potassium, Soil Biol. Biochem. 16, 589–594.
- Macduff J.H., Jarvis S.C., Davidson I.A. (1996) Inhibition of  $N_2$  fixation by white clover (*Trifolium repens* L.) at low concentrations of  $NO_3^-$  in flowing solution culture, Plant Soil 180, 287–295.
- Mahon J.D. (1977a) Respiration and the energy requirement for nitrogen fixation in nodulated pea roots, Plant Physiol. 60, 817–821.
- Mahon J.D. (1977b) Root and nodule respiration in relation to acetylene reduction in intact nodulated peas, Plant Physiol. 60, 812–816.
- Malézieux E., Crozat Y., Dupraz C., Laurans M., Makowski D., Ozier-Lafontaine H., Rapidel B., de Tourdonnet S., Valantin-Morison M. (2009) Mixing plant species in cropping systems: concepts, tools and models. A review, Agron. Sustain. Dev. 29, 43–62.
- Masterson C.L., Murphy P.M. (1976) Application of the acetylene reduction technique to the study of nitrogen fixation

by white clover in the field, in: Nutman P.S. (Eds.), Symbiotic Nitrogen Fixation in Plants, Cambridge University Press, London, UK, pp. 299–316.

- McCallum M.H., Peoples M.B., Connor D.J. (1999) Contributions of nitrogen by field pea (*Pisum sativum* L.) in a continuous cropping sequence compared with a lucerne (*Medicago sativa* L.)-based pasture ley in the Victorian Wimmera, Aust. J. Agric. Res. 51, 13–22.
- Michiels J., Verreth C., Vanderleyden J. (1994) Effects of temperature stress on bean-nodulating *Rhizobium* strains, Appl. Environ. Microbiol. 60, 1206–1212.
- Minchin F.R., Pate J.S. (1973) The carbon balance of a legume and the functional economy of its root nodules, J. Exp. Bot. 24, 259–271.
- Minchin F.R., Summerfield R.J., Hadley P., Roberts E.H., Rawsthorne S. (1981) Carbon and nitrogen nutrition of nodulated roots of grain legumes, Plant Cell Environ. 4, 5–26.
- Munns D. (1970) Nodulation of *Medicago sativa* in solution culture V. calcium and pH requirements during infection, Plant Soil 32, 90–102.
- Nelson L.M., Child J.J. (1981) Nitrogen-fixation and hydrogen metabolism by *Rhizobium-leguminosarum* isolates in pea root nodules, Can. J. Microbiol. 27, 1028–1034.
- Nelson D.R., Bellville R.J., Porter C.A. (1984) Role of nitrogen assimilation in seed development of soybean, Plant Physiol. 74, 128–133.
- Pate J.S. (1961) Temperature characteristics of bacterial variation in legume symbiosis, Nature 192, 637–639.
- Peoples M.B., Lilley D.M., Burnett V.F., Ridley A.M., Garden D.L. (1995) Effects of surface application of lime and superphosphate to acid soils on growth and N<sub>2</sub> fixation by subterranean clover in mixed pasture swards, Soil Biol. Biochem. 27, 663–671.
- Phillips D.A. (1980) Efficiency of symbiotic nitrogen fixation in legumes, Annu. Rev. Plant Physiol. 31, 29–49.
- Piha M.I., Munns D.N. (1987) Sensitivity of the common bean (*Phaseolus vulgaris* L.) symbiosis to high soil temperature, Plant Soil 98, 183–194.
- Pimratch S., Jogloy S., Vorasoot N., Toomsan B., Patanothai A., Holbrook C.C. (2008) Relationship between biomass production and nitrogen fixation under drought-stress conditions in peanut genotypes with different levels of drought resistance, J. Agron. Crop Sci. 194, 15–25.
- Postgate J.R. (1982) Biological nitrogen fixation: fundamentals, Philos. Trans. R. Soc. B 296, 387–375.
- Purcell L.C., Sinclair T.R. (1990) Nitrogenase activity and nodule gas permeability response to rhizospheric NH<sub>3</sub> in soybean, Plant Physiol. 92, 268–272.
- Purcell L.C., de Silva M., King C.A., Kim W.H. (1997) Biomass accumulation and allocation in soybean associated with genotypic differences in tolerance of nitrogen fixation to water deficits, Plant Soil 196, 101–113.
- Rao V.R. (1977) Effect of root temperature on the infection processes and nodulation in *Lotus* and *Stylosanthes*, J. Exp. Bot. 28, 241–259.
- Rechcigl J.E., Edmisten K.L., Wolf D.D., Reneau R.B. Jr. (1987) Influence of calcium, nitrogen, and pH on alfalfa root growth and nitrogen fixation using the implanted soil mass technique, Agron. J. 79, 926–928.

- Rice W.A., Clayton G.W., Olsen P.E., Lupwayi N.Z. (2000) Rhizobial inoculant formulations and soil pH influence field pea nodulation and nitrogen fixation, Can. J. Soil Sci. 80, 395–400.
- Richardson A.C., Syers J.K. (1985) Edaphic limitations and soil nutrient requirements of legume-based forage systems in temperate regions of New Zealand, in: Barnes R.F., Ball P.R., Broughman R.W., Marten G.C., Minson D.J. (Eds.), Forage Legumes for Energy-efficient Animal Production, USDA, Washington, pp. 89–94.
- Robertson M.J., Carberry P.S., Huth N.I., Turpin J.E., Probert M.E., Poulton P.L., Bell M., Wright G.C., Yeates S.J., Brinsmead R.B. (2002) Simulation of growth and development of diverse legume species in APSIM, Aust. J. Agric. Res. 53, 429–446.
- Roughley R.J. (1970) The influence of root temperature, *Rhi-zobium* strain and host selection on the structure and nitrogen-fixing efficiency of the root nodules of *Trifolium subterraneum*, Ann. Bot. 34, 631–646.
- Roughley R.J., Dart P.J. (1969) Reduction of acetylene by nodules of *Trifolium subterraneum* as affected by root temperature, *Rhizobium* strain and host cultivar, Arch. Microbiol. 69, 171–179.
- Roughley R., Dart P. (1970) Root temperature and root-hair infection of *Trifolium subterraneum* L. cv. Cranmore, Plant Soil 32, 518–520.
- Ryle G.J.A., Powell C.E., Gordon A.J. (1979) The respiratory costs of nitrogen fixation in soyabean, cowpea, and white clover I. nitrogen fixation and the respiration of the nodulated root, J. Exp. Bot. 30, 135–144.
- Sall K., Sinclair T.R. (1991) Soybean genotypic differences in sensitivity of symbiotic nitrogen fixation to soil dehydration, Plant Soil 133, 31–37.
- Sau F., Boote K.J., Ruíz-Nogueira B. (1999) Evaluation and improvement of CROPGRO-soybean model for a cool environment in Galicia, northwest Spain, Field Crops Res. 61, 273–291.
- Schiltz S., Munier-Jolain N., Jeudy C., Burstin J., Salon C. (2005) Dynamics of exogenous nitrogen partitioning and nitrogen remobilization from vegetative organs in pea revealed by <sup>15</sup>N in vivo labeling throughout seed filling, Plant Physiol. 137, 1463–1473.
- Schmid M., Neftel A., Riedo M., Fuhrer J. (2001) Process-based modelling of nitrous oxide emissions from different nitrogen sources in mown grassland, Nutr. Cycl. Agroecosyst. 60, 177–187.
- Schomberg H.H., Weaver R.W. (1992) Nodulation, nitrogen fixation, and early growth of arrowleaf clover in response to root temperature and starter nitrogen, Agron. J. 84, 1046–1050.
- Schulze J., Adgo E., Merbach W. (1999) Carbon costs associated with N<sub>2</sub> fixation in *Vicia faba* L. and *Pisum sativum* 1. over a 14-day period, Plant Biol. 1, 625–631.
- Schwinning S., Parsons A.J. (1996) Analysis of the coexistence mechanisms for grasses and legumes in grazing systems, J. Ecol. 84, 799–813.
- Serraj R., Drevon J.J. (1998) Effects of salinity and nitrogen source on growth and nitrogen fixation in alfalfa, J. Plant Nutr. 21, 1805–1818.

- Serraj R., Sinclair T.R. (1998) N<sub>2</sub> fixation response to drought in common bean (*Phaseolus vulgaris* L.), Ann. Bot. 82, 229–234.
- Serraj R., Bona S., Purcell L.C., Sinclair T.R. (2001) Role of soil-oxygen enrichment in the response of nitrogen fixation to drought stress in field-grown soybean, Soil Crop Sci. Soc. Fla. Proc. 60, 88–93.
- Sharpley A.N., Williams J.R. (1990) EPIC-erosion/productivity impact calculator 1. Model documentation, USDA Tech. Bull. No. 1768, pp. 1–235.
- Sheehy J.E. (1987) Photosynthesis and nitrogen fixation in legume plants, Crit. Rev. Plant Sci. 5, 121–159.
- Sheehy J.E., Woodward F.I., Gosse G. (1991) Measurements of nitrogen fixation ( $C_2H_2$ ), photosynthesis and respiration using an open system in the natural environment, Ann. Bot. 67, 131–136.
- Sinclair T.R. (1986) Water and nitrogen limitations in soybean grain production I. model development, Field Crops Res. 15, 125–141.
- Sinclair T.R., Muchow R.C., Ludlow M.M., Leach G.J., Lawn R.J., Foale M.A. (1987) Field and model analysis of the effect of water deficits on carbon and nitrogen accumulation by soybean, cowpea and black gram, Field Crops Res. 17, 121–140.
- Small J.G.C., Joffe A. (1968) Physiological studies on the genus *Trifolium* with special reference to the South African species. II. Influence of root temperature on growth, nodulation and symbiotic nitrogen fixation, S. African J. Agric. Sci. 11, 41–56.
- Soussana J.F., Minchin F.R., Macduff J.H., Raistrick N., Abberton M.T., Michaelson-Yeates T.P.T. (2002) A simple model of feedback regulation for nitrate uptake and N<sub>2</sub> fixation in contrasting phenotypes of white clover, Ann. Bot. 90, 139–147.
- Sprent J.I. (1972) Effects of water stress on nitrogen-fixing root nodules. IV. Effects on whole plants of *Vicia Faba* and *Glycine max*, New Phytol. 71, 603–611.
- Sprent J.I. (1976) Water deficits and nitrogen-fixing root nodules, in: Kozlowski T.T. (Ed.), Water Deficits and Plant Growth Vol IV. Soil water measurement, plant responses, and breeding for drought resistance. Academic Press, New York, pp. 291–315.
- Stout W.L., Fales S.L., Muller L.D., Schnabel R.R., Weaver S.R. (2000) Water quality implications of nitrate leaching from intensively grazed pasture swards in the northeast US, Agric. Ecosyst. Environ. 77, 203–210.
- Streeter J. (1988) Inhibition of legume nodule formation and  $N_2$  fixation by nitrate, Crit. Rev. Plant Sci. 7, 1–23.
- Svenning M.M., Junttila O., Macduff J.H. (1996) Differential rates of inhibition of N<sub>2</sub> fixation by sustained low concentrations of  $NH_4^+$  and  $NO_3^-$  in northern ecotypes of white clover (*Trifolium repens* L.), J. Exp. Bot. 47, 729–738.
- Thornley J.H.M. (1998) Plant Submodel, in: Thornley J.H.M. (Ed.), Grassland Dynamics: An Ecosystem Simulation Model, CAB International, Wallingford, Oxon, UK, pp. 53–57.
- Thornley J.H.M. (2001) Simulating grass-legume dynamics: a phenomenological submodel, Ann. Bot. 88, 905–913.
- Thornley J.H.M., Cannell M.G.R. (2000) Dynamics of mineral N availability in grassland ecosystems under increased

[CO<sub>2</sub>]: hypotheses evaluated using the Hurley Pasture Model, Plant Soil 224, 153–170.

- Thornley J.H.M., Bergelson J., Parsons A.J. (1995) Complex dynamics in a carbon-nitrogen model of a grass-legume pastures, Ann. Bot. 75, 79–94.
- Trindade H., Coutinho J., Jarvis S., Moreira N. (2001) Nitrogen mineralization in sandy loam soils under an intensive doublecropping forage system with dairy-cattle slurry applications, Eur. J. Agron. 15, 281–293.
- Twary S.N., Heichel G.H. (1991) Carbon costs of dinitrogen fixation associated with dry matter accumulation in alfalfa, Crop Sci. 31, 985–992.
- Venkateswarlu B., Saharan N., Maheswari M. (1990) Nodulation and N<sub>2</sub> (C<sub>2</sub>H<sub>2</sub>) fixation in cowpea and groundnut during water stress and recovery, Field Crops Res. 25, 223–232.
- Voisin A.-S., Bourion V., Duc G., Salon C. (2007) Using an ecophysiological analysis to dissect genetic variability and to propose an ideotype for nitrogen nutrition in pea, Ann. Bot. 100, 1525–1536.
- Voisin A.-S., Salon C., Jeudy C., Warembourg F.R. (2003) Symbiotic N<sub>2</sub> fixation activity in relation to C economy of *Pisum sativum* L. as a function of plant phenology, J. Exp. Bot. 54, 2733–2744.
- Voisin A.-S., Salon C., Munier-Jolain N.G., Ney B. (2002) Effect of mineral nitrogen on nitrogen nutrition and biomass partitioning between the shoot and roots of pea (*Pisum sativum* L.), Plant Soil 242, 251–262.
- Warembourg F.R., Roumet C. (1989) Why and how to estimate the cost of symbiotic N<sub>2</sub> fixation? A progressive approach based on the use of <sup>14</sup>C and <sup>15</sup>N isotopes, Plant Soil 115, 167–177.
- Waterer J.G., Vessey J.K. (1993a) Effect of low static nitrate concentrations on mineral nitrogen uptake, nodulation, and nitrogen fixation in field pea, J. Plant Nutr. 16, 1775–1789.

- Waterer J.G., Vessey J.K. (1993b) Nodulation response of autoregulated or NH<sup>+</sup><sub>4</sub>-inhibited pea (*Pisum sativum*) after transfer to stimulatory (low) concentrations of NH<sup>+</sup><sub>4</sub>, Physiol. Plant. 88, 460–466.
- Waterer J.G., Vessey J.K., David Raper J.C. (1992) Stimulation of nodulation in field peas (*Pisum sativum*) by low concentrations of ammonium in hydroponic culture, Physiol. Plant. 86, 215–220.
- Watson C.A., Goss M.J. (1997) Estimation of N<sub>2</sub>-fixation by grass-white clover mixtures in cut or grazed swards, Soil Use Manage. 13, 165–167.
- Waughman G.J. (1977) The effect of temperature on nitrogenase activity, J. Exp. Bot. 28, 949–960.
- Weisz P.R., Sinclair T.R. (1987) Regulation of soybean nitrogen fixation in response to rhizosphere oxygen. II. Quantification of nodule gas permeability, Plant Physiol. 84, 906–910.
- Weisz P.R., Denison R.F., Sinclair T.R. (1985) Response to drought stress of nitrogen fixation (acetylene reduction) rates by field-grown soybeans, Plant Physiol. 78, 525–530.
- Whitehead D.C. (1995) Legumes: Biological Nitrogen Fixation and Interaction with Grasses, in: Whitehead D.C. (Ed.), Grassland Nitrogen, CAB International, Wallingford, UK, pp. 36–57.
- Wood M. (1996) Nitrogen Fixation: How much and at what cost? in: Younie D. (Ed.), Legumes in Sustainable Farming Systems, Occasional Symposim No. 30, British Grassland Society, Reading, UK, pp. 26–35.
- Wu L., McGechan M.B. (1999) Simulation of nitrogen uptake, fixation and leaching in a grass/white clover mixture, Grass Forage Sci. 54, 30–41.
- Yu M., Gao Q., Shaffer M.J. (2002) Simulating interactive effects of symbiotic nitrogen fixation, carbon dioxide elevation, and climatic change on legume growth, J. Environ. Qual. 31, 634–641.
# **Arbuscular Mycorrhizal Networks: Process and Functions**

Neera Garg and Shikha Chandel

### Contents

1	Intro	oduction	908		
2	A Journey Through Mycorrhizal Symbiosis				
	2.1	Arbuscular Mycorrhizal Development	910		
	2.2	Plant Signal and Fungal Perception	911		
3	Мус	orrhizae and Their Diverse Roles	915		
	3.1	Mycorrhizal Symbiosis and Mineral Nutrition .	915		
	3.2	Alleviation of Salt Stress by Arbuscular			
		Mycorrhizal (AM) Fungi	918		
	3.3	Plant Water Relationship	919		
	3.4	Protection of Host Roots from Pathogens	919		
	3.5	Mycorrhizoremediation	921		
	3.6	Miscellaneous Roles of Arbuscular			
		Mycorrhizal (AM) Fungi	922		
4	Con	clusion	923		
Re	eferen	ces	923		

N. Garg (🖂)

Department of Botany, Panjab University, Chandigarh – 160014, India e-mail: garg\_neera@yahoo.com Abstract An unprecedented, rapid change in environmental conditions is being observed, which invariably overrules the adaptive capacity of land plants. These environmental changes mainly originate from anthropogenic activities, which have aggravated air and soil pollution, acid precipitation, soil degradation, salinity, contamination of natural and agro-ecosystems with heavy metals such as cadmium (Cd), lead (Pb), mercury (Hg), arsenic (As), global climate change, etc. The restoration of degraded natural habitats using sustainable, low-input cropping systems with the aim of maximizing yields of crop plants is the need of the hour. Thus, incorporation of the natural roles of beneficial microorganisms in maintaining soil fertility and plant productivity is gaining importance and may be an important approach. Symbiotic association of the majority of crop plants with arbuscular mycorrhizal (AM) fungi plays a central role in many microbiological and ecological processes. In mycorrhizal associations, the fungal partner assists its plant host in phosphorus (P) and nitrogen (N) uptake and also some of the relatively immobile trace elements such as zinc (Zn), copper (Cu) and iron (Fe). AM fungi also benefit plants by increasing water uptake, plant resistance and biocontrol of phytopathogens, adaptation to a variety of environmental stresses such as drought, heat, salinity, heavy metal contamination, production of growth hormones and certain enzymes, and even in the uptake of radioactive elements. The establishment of symbiotic association usually involves mutual recognition and a high degree of coordination at the morphological and physiological level, which requires a continuous cellular and molecular dialogue between both the partners. This has led to the identification of the genes, signal transduction pathways and the chemical structures of components relevant to symbiosis; however, scientific knowledge on the physiology and function of these fungi is still limited. This review unfolds our current knowledge on signals and mechanisms in the development of AM symbiosis; the molecular basis of nutrient exchange between AM fungi and host plants; and the role of AM fungi in water uptake, disease protection, alleviation of various abiotic soil stresses and increasing grain production.

**Keywords** Arbuscular mycorrhiza • Environmental stresses • Phytopathogens • Sustainable agriculture

### 1 Introduction

There is a pressing need to strike a balance between food production for the ever-increasing world population and conserving biodiversity, as well as supply of organic or low-input products for the markets (Dodd, 2000). Earlier, soil was treated as a 'resource base' with its function of support subordinate to the production of food and fiber, and on the scale of priorities, the soil had always taken second place (Bethlenfalvay and Schüepp, 1994). Recently, scientists have recognized the importance of the soil not only as an agricultural resource base (Stewart et al., 1991), but as a complex, living and fragile system that must be protected (Reagnold et al., 1990) and managed for its own sake (Pierce and Lal, 1991) to guarantee its long-term stability and productivity (Aryal and Xu, 2001).

Increased environmental awareness has progressively led to a shift from conventional intensive management to low-input, sustainable crop production agroecosystems. In low-input cropping systems the natural activities of microorganisms contribute to the biocontrol of plant pathogens and improved supply of nutrients, thus maintaining crop health and production. Symbiotic mycorrhizal fungi, such as arbuscular mycorrhizal (AM) fungi, form a key component of the microbial populations influencing plant growth and soil productivity (Johansson et al., 2004). The multifunctional nature of AM fungi include weathering, dissolution and cycling of mineral nutrients (Finlay and Rosling, 2006; Wallander, 2006; Helgason and Fitter, 2009), nutrient mobilization from organic substrates (Finlay, 2008), carbon cycling (Johnson et al.,

2002), effects on plant communities and ecosystems, and mediation of plant responses to various environmental stresses, such as soil salinity, heavy metal toxicity, drought and heat stress, soil acidification, and plant pathogens (Colpaert, 2008; Finlay et al., 2008), as well as a range of possible interactions with groups of other soil microbes (Finlay, 2008). The natural roles of these microorganisms may have been marginalized in intensive agriculture, since microbial communities in conventional farming systems have been modified due to tillage (Sturz et al., 1997; McGonigle and Miller, 1996) and high inputs of inorganic fertilizers, herbicides and pesticides (Gianinazzi and Schüepp, 1994; Gianinazzi et al., 2002). A better understanding of the microbial interactions is therefore crucial for the development of sustainable management of soil fertility and crop production.

The vast majority of plant species in terrestrial ecosystems form symbioses with rhizosphere microbes to take up essential nutrients (Bonfante and Genre, 2008; Parniske, 2008; Helgason and Fitter, 2009). A number of these microbes inhabit the rhizosphere, including mycorrhizal fungi, nitrogen-fixing bacteria and other plant-growth-promoting rhizobacteria (van der Putten et al., 2007). The endosymbiosis formed between roots of more than 80% of land plant species and arbuscular mycorrhizal (AM) fungi is ubiquitous and the most widespread symbiotic association in the plant kingdom (Redecker et al., 2000; Helgason and Fitter, 2009). The AM fungi were formerly included in the order Glomales in the Zygomycota (Redecker et al., 2000; Bonfante and Genre, 2008), but they have recently been moved to a newly ascribed phylum, Glomeromycota (Garg et al., 2006; Parniske, 2008; Smith and Read, 2008). Arbuscular mycorrhizal fungi develop extensive, below-ground extraradical hyphae fundamental for the uptake of inorganic phosphate and other immobile nutrients from the soil and their translocation to the host plant (Giovannetti et al., 2006; Smith and Read, 2008). Fungal penetration and establishment in the host roots involve a complex sequence of events and intracellular modifications (Bonfante-Fasolo and Perotto, 1992). In contrast, the nitrogenfixing root nodule symbiosis is almost completely restricted to the legumes (Zhu et al., 2006). The symbiotic relationships between legumes and rhizobacteria involve extensive signaling between the two organisms (Smit et al., 2005). These microbes can stimulate plant

growth and reproduction by providing their hosts with services such as increased access to limiting nutrients (phosphorus and nitrogen) and enhanced uptake of water. In exchange, the plants provide these microbes with carbon (Parniske, 2008; Smith and Read, 2008). An important characteristic of legumes is that they utilize more nitrogen and phosphorus from the environment with the help of nodulation bacteria and mycorrhizal fungi living in their roots (Smith and Read, 2008). Thus, dual application of AM fungi and *Rhizobium* acts as biological fertilization in legumes. Moreover, there are ameliorative synergistic effects of this dual application (Abd-Alla et al., 2000).

## 2 A Journey Through Mycorrhizal Symbiosis

The term 'mycorrhiza' originates from the Greek *mycos*, meaning 'fungus', and *rhiza* meaning 'root', and was first used in 1885 (Frank, 1885) to describe the intimate association between biotrophic mycorrhizal fungi and plant roots. Approximately 80% of land plant species that have been studied form the

mycorrhizal symbiosis (and 92% of plant families), which exist everywhere, from tiny home-gardens to large ecosystems (Wang and Qiu, 2006; Helgason and Fitter, 2009). Six types of mycorhizas; arbuscular, arbutoid, ecto-, ericoid, monoptropoid and orchid, are categorized by their distinct morphological characteristics (Wang and Qiu, 2006; Garg et al., 2006). Of them, arbuscular mycorrhiza (AM) is the most common and predominant type (Fig. 1). Arbuscules, specific 'littletree-shaped' fungal structures, serve as the main sites of nutrient exchange between the plant and the fungus (He and Nara, 2007). They are branched, microscopic haustorial structures of the fungal symbiont that form within living cortical cells of the root (Manchanda and Garg, 2007). This structure is common to all associations of this type of mycorrhiza (Franken et al., 2007). AM fungi are soil inhabitants with a presumed origin at least 460 million years ago (Redecker et al., 2000; Bonfante and Genre, 2008). Perhaps due to this ancient association with plants, AM fungi have lost their ability to live and complete their life cycle in the absence of a green partner (Requena et al., 2007). The Glomeromycota consists of approximately 150 isolates which colonize a wide range of both monoand dicotyledonous plant species (Paszkowski, 2006).





These AM fungi communities influence a number of important ecosystem processes, including plant productivity, plant diversity and soil structure (Van der Heijden et al., 1998). So, not only do the activities of AM fungi have multiple functions that enhance plant performance, but they also play crucial roles in the development of soil properties and the health of the entire ecosystem.

### 2.1 Arbuscular Mycorrhizal Development

The establishment of AM symbiosis can be envisaged as a programed sequence of phenotypic changes, corresponding to distinct recognition events which lead the two partners, host plant and fungal symbiont, to a high degree of morphological and physiological integration.

### 2.1.1 Early Stages of Fungal Asymbiotic Growth

Arbuscular mycorrhizal fungi are obligate biotrophs, unable to complete their life cycle during asymbiosis (Bonfante and Bianciotto, 1995). AM fungal spores are the only plant-independent phase of the mycobiont. They are round-shaped structures with a thick cell wall and average diameters between 50 and 100 µm. They contain a very large number of nuclei, up to 2000 per spore (Bécard and Pfeffer, 1993). After germination, hyphae are always coenocytic. Studies on two AM fungal species have shown that these are haploids with an unusually high genetic variation (Hijri and Sanders, 2004; Hosny et al., 1997). Assessments of the genome size of these fungi have shown extreme variations between different species ranging from about 16.5 Mb in Glomus intraradices (Hijri and Sanders, 2004) up to 1058.4 Mb in Scutellospora gregaria (Hosny et al., 1998). AM spores germinate under appropriate water and temperature conditions and hyphae grow for about 2-3 weeks. Several nuclei from the spore move into the extending mycelium and some of them undergo mitosis (Requena et al., 2000). During this time the fungal colonies extend a few centimeters, showing a characteristic growth pattern with marked apical dominance and infrequent hyphal branching. In the absence of a host root, growth ceases after about 2-4 weeks and hyphal septation from the apex occurs (Mosse, 1988).

The apical septation is accompanied by extensive vacuolization and retraction of protoplasm towards the spore (Logi et al., 1998). During this asymbiotic phase, the fungus lives mainly on its triacylglyceride reserves. This phase of growth in the absence of signals from the plant is what is known as the asymbiotic stage (Requena et al., 2007). Thus, in the asymbiotic stage, spores germinate and AM fungi show limited hyphal development in the absence of a host plant.

#### 2.1.2 Arbuscular Mycorrhizal Life Cycle

The establishment of the AM symbiosis begins with the colonization of a compatible root by the hyphae produced by AM fungal soil propagules, asexual spores or mycorrhizal roots (Requena et al., 1996). After attachment of a hypha to the root surface by means of an appressorium, the fungus penetrates into the cortex and forms distinct morphologically specialized structures: inter- and intracellular hyphae, coils and arbuscules. Arbuscules are specialized hyphae, similar to haustoria from the plant pathogenic fungi, formed as intercalary structures between the coil hyphae, and are the site of mineral nutrient transfer to the plant and potentially the site of carbon acquisition by the fungus (Requena et al., 2007; Pumplin and Harrison, 2009). After host colonization, the fungal mycelium grows out of the root exploring the soil in search of mineral nutrients, and it can colonize other susceptible roots. The fungal life cycle is completed after formation of asexual chlamydospores on the external mycelium. Distinct morphological stages can therefore be identified during the life cycle of arbuscular mycorrhizal fungi (Requena and Breuninger, 2004). This clearly shows that the host plant plays a key role in orchestrating the AM infection process (Eckardt, 2005).

The sequence of steps leading to an AM symbiosis is largely conserved among different combinations of fungal and plant species. Overall, these developmental processes require molecular communication between the AM fungus and the plant, including exchange and perception of signals by the symbiotic partners (Bucher, 2007). Thus, the complex morphological and physiological alterations of both symbiotic partners accompanied by the recognition process suggest that AM symbiosis is the result of multifaceted, fine-tuned signaling events (Paszkowski, 2006).

### 2.2 Plant Signal and Fungal Perception

This phase of mycorrhizal fungi can be divided into three steps.

#### 2.2.1 Pre-symbiotic Phase

Fungal Responses to Plant-Derived Signals

For both symbionts, the period before physical contact (appressorium formation) involves recognition and attraction of appropriate partners and other events promoting an alliance. There is increasing evidence showing that the fungus and plant start to recognize each other long before the first colonization structures on the root epidermis appear (Requena et al., 2007). Spores of AM fungi persist in the soil and germinate spontaneously, independently of plant-derived signals. However, root exudates and volatiles may promote or suppress spore germination, indicating the existence of the presence of spore 'receptors' responsive to alterations in the chemical composition of the environment (Bécard et al., 2004; Harrison, 2005). It has been known for a long time that germinating hyphae from spores respond to the presence of roots in their vicinity. In the vicinity of a host root, fungal morphology shifts towards enhanced hyphal growth and extensive hyphal branching (Giovannetti et al., 1993b; Buee et al., 2000). Although no directional growth has been observed towards the root, several experiments showed that exudates from the host root elicit growth stimulation in contrast to non-host root exudates (Giovannetti et al., 1993a,b, 1996). These observations suggest that the fungus senses a host-derived signal; ('branching factor'), leading to intensified hyphal ramification that is likely to increase the probability of contact with a host root. Hence, distinction between host and nonhost occurs to a certain degree at this early point in the interaction (Paszkowski, 2006).

In many plant-microbe interactions, the dialogue between the two symbionts is initiated by the presence of plant phenolic substances such as flavonoids. Interestingly, there are many reports showing that, indeed, flavonoids (exogenously applied to spores) exert a positive effect on hyphal growth during asymbiosis (Akiyama et al., 2002). However, flavonoids might not be essential for the plant-fungal recognition since a study using maize mutant plants impaired in flavonoid production showed that they were able to form mycorrhizal symbiosis similarly to wild-type plants (Bécard et al., 1995). A major step forward in deciphering the molecular cross-talk in the AM symbiosis was the identification of a branching factor present in root exudates as a strigolactone (5-deoxystrigol) (Akiyama et al., 2005). Strigolactones have been isolated from a wide range of mono- and dicotyledonous plants and were previously found to stimulate seed germination of parasitic weeds such as Striga and Orobranche (Bouwmeester et al., 2003). Strigolactones were previously described as sesquiterpenes (Akiyama et al., 2005); however, the use of carotenoid mutants of maize and inhibitors of isoprenoid pathways in maize, sorghum and cowpea showed that strigolactones are derived from the carotenoid biosynthetic pathway (Matusova et al., 2005). Nevertheless, the results do indicate that processed carotenoid derivatives are involved at multiple stages in the development of the AM symbiosis, possibly by stimulating intraradical fungal branching (Paszkowski, 2006). It is likely that other signals, such as thigmotrophic signals from the plant surface or secondary metabolites produced in plants after perception of the fungus, are required for appressorium formation and symbiosis progression (Requena et al., 2007). These chemical signals exuded by the plant and the thigmotrophic signals from the rhizodermis are possibly recognized by receptor proteins associated with the fungal plasma membrane (Requena et al., 2007). Requena et al. (2002) used suppressive subtractive hybridization (SSH) to create a subtractive cDNA library from Glomus mosseae enriched in genes induced during the asymbiotic phase. With this approach, a novel gene (GmGin1) encoding a two-domain protein with a putative role in signaling was identified. Expression analyses showed that GmGin1 was downregulated upon entry into symbiosis, suggesting it could play a role at the plant recognition stage (Requena et al., 2002). GmGin1 could be a sensor for plant signals and is located at the cell membrane. It undergoes splicing in response to signals from the plant. These interesting results show that the chemical communication with the plant symbiont not only modifies fungal gene expres-

sion but it is also able to induce post-transcriptional modification of fungal proteins (Requena et al., 2007). Transcript profiling of AM fungi during appressorium formation showed that plant contact induces the activation of genes from different categories, including

several components related to Ca<sup>2+</sup>-signaling, including a putative P-type Ca<sup>2+</sup>-ATPase, a calmodulin, a leucine zipper EF-protein, and a Ca<sup>2+</sup>-induced Ras inactivator (CAPRI) (Breuninger and Requena, 2004). This suggests the involvement of  $Ca^{2+}$  as a second messenger in the transmission of plant-derived signals, leading to appressorium formation in the AM symbiosis (Requena et al., 2007). Thus, upon detection of a host root there is vigorous investment in the production of fungal hyphae, which can then rapidly make contact with essential carbon sources. A major breakthrough in the molecular interactions between both the partners was the identification of the host branching factor 5deoxy-strigol, which induces the so-called presymbiotic stage characterized by continued hyphal growth, increased physiological activity and profuse branching of hyphae, and the gene, GmGin1, which plays a role in the development of appressoria during AM symbiosis in the plants.

### Plant Responses to Fungus-Derived Signals

The plant responds to the microbial profile of the rhizosphere in different ways depending upon the type of organism present. Detection of pathogen-derived elicitors triggers plant signaling cascades that lead to a defense response (Glazebrook, 2005). On the other hand, plant defense responses are either not mounted at all or mounted only transiently before being suppressed during AM symbiosis (Harrison, 2005). Extensive forward genetic approaches were used to dissect the components of the signal perception and transduction pathways in the AM symbiosis in legume species. These approaches led to the identification of several symbiosis genes, the corresponding plant mutants of which are generally unable to support infection by AM fungi. Yet the nature and function of mycorrhization (Myc) factors are still an enigma. Myc factors are likely to be soluble, fungus-derived compounds that trigger expression of mycorrhizalresponsive genes (Kosuta et al., 2003) and structural changes in host roots (Olah et al., 2005).

GUS reporter expression [*EARLY NODULA-TION11 (ENOD11*)-promoter:  $\beta$ -glucuronidase], which is responsive to both AM fungi and a rhizobial Nod-factor, before contact was monitored in root sections adjacent to intensely branching hyphae. The

intensity and distribution of GUS expression indicated the detection by the plant of a compound released by the fungus (Kosuta et al., 2003). Interestingly, when contact and penetration were permitted, GUS expression was restricted to infected and associated cells (Chabaud et al., 2002; Genre et al., 2005), indicating the induction of a suppressor activity in non-colonized neighboring cells (Parniske, 2004). When contact is made and a precise area of penetration and colonization is established, gene expression is restricted to cells in direct contact with the penetrating fungus. Similarly, during preinfection and infection stages in the interaction of Medicago truncatula with nitrogen-fixing Rhizobia, GUS expression was found in the rhizodermis of the larger root section before contact and upon subsequent bacterial entrance into the root tissue became confined to the area of infection, namely the infection site and the invaded nodule (Journet et al., 2001). This resemblance in expression patterns between the two types of root symbiosis suggests that the responses are a part of a rather general symbiont 'anticipation' program (Paszkowski, 2006). To conclude, the infected cells in the plant roots upon perceiving Myc factors trigger the expression of SYM genes (ENOD11-promoter GUS reporter gene), common to both bacterial and fungal root endosymbioses, providing a glimpse of conservation and specialization of signaling cascades essential for nodulation and mycorrhiza development.

#### 2.2.2 Early Symbiotic Phase

#### Appressorium Development

The onset of the symbiosis is marked morphologically by the formation of appressoria, the cell-to-cell contact between the fungus and plant and the site of fungal ingress into the host root. The formation of appressoria is one of the first morphological signs that recognition between the plant and the fungus has occurred (Garcio-Garrido and Ocampo, 2002). The development of appressoria can be considered to be the result of successful presymbiotic recognition events when fungal and plant partners are committed to an interaction (Giovannetti et al., 1993a). Structurally, appressoria differ from hyphae by being flattened, elliptical hyphal tips that adhere by unknown means to the surface of host rhizodermal cells. This morphological switch is reflected by changes in fungal gene transcription (Breuninger and Requena, 2004). While physical and chemical rhizodermal cell wall features are required and are sufficient to elicit appressorium formation, penetration needs the support of intact cells whose coordinated, matching response accommodates the fungus (Paszkowski, 2006). Therefore, the discovery that the plant cell actively prepares the intracellular environment for AM fungal hyphae elicited its role during the infection process, and emphasized the significance and indispensable participation of plant processes in the coordinated invasion (Genre et al., 2005, 2008).

#### Arbuscular Mycorrhizal (AM) Fungi Penetration

Actual penetration of the host by AM fungi is by means of a penetration hyphae or infection peg from the appressorium that penetrates either by force or by production of cell wall-dissolving enzymes (Linderman, 1994). Specific plant-regulated processes associated with penetration have been delineated. In Lotus japo*nius*, for example, an epidermal cleft opens between two adjacent rhizodermal cells through which the fungus enters. From here, invasion of the rhizodermal cells is initiated and the fungus trespasses through the underlying exodermal cells (Parniske, 2004). In addition to entering through a rhizodermal cleft, the fungus can also traverse rhizodermal cells directly, as shown in detail for Gigaspora gigantea penetrating roots of Medicago truncatula (Genre et al., 2005). Hence, infection only occurs after preparatory activities in the plant cell. These cytological studies illustrate nicely the extensive complementary contribution made by host cell activity to fungal penetration. This emphasizes the significance and indispensable participation of plant processes in the coordinated invasion (Genre et al., 2005, 2008).

The infection process by arbuscular mycorrhizal fungi is also characterized by a low and regulated production of cell wall-degrading enzymes by the fungus. The production of exo- and endoglucanases, cellulases, xyloglucanases and pectolytic enzymes including polygalacturonase has been demonstrated in various investigations (Garcia-Garrido and Ocampo, 2002). In conclusion, AM fungi seem to colonize the root tissues of host plant by means of a combination of mechanical and enzymatic mechanisms and the coordinated participation of host cell machinery, which aids in the easier penetration of the mycorrhizal fungi into the root cortex (Bonfante and Perotto, 1992).

### 2.2.3 Mature Symbiotic Phase: Process and Signaling

#### Arbuscule Development

There are two main morphological types (Paris and Arum) of AM symbioses with structurally different interfaces. Paris-type AM are characterized by intracellular fungal coils that grow directly from cell to cell with little or no intercellular phase, whereas in Arumtype AM, a highly branched intracellular tree-like fungal structure known as the arbuscule is formed, subtended by intercellular hyphae (Dickson et al., 2007). Arbuscules are the key feature of the AM symbiosis as they represent an extreme form of intimacy and compatibility and are thought to be the site of nutrient transfer from the fungus to the host plant (Hughes et al., 2008). Following colonization of the host cell by the arbuscule, the architecture of the host cell undergoes remarkable changes. Despite the intense activity of both the partners leading towards arbusculated cells, arbuscules collapse after 2-4 days, leaving an intact cortical cell that is then able to host another arbuscule (Paszkowski, 2006; Pumplin and Harrison, 2009).

It is not known what triggers fungal entrance into the host cell but the perception of a radical sugar gradient between the vascular tissue and the outer cell layers may be involved in induction of arbuscule formation (Blee and Anderson, 1998). Alternatively, the initiation of arbuscule collapse may be caused by endogenous fungal signaling or coordinated signaling cross-talk. Development of arbuscules is at least partially under the control of the host genetic program. In addition, proteins of the SYM pathway are either required for arbuscule development, such as *LjCASTOR*, LjSYM15 and likely LjSYm6, or contribute to arbuscule formation, such as LjPOLLUX, LjNup 133 and LjSYM 24 (Kistner et al., 2005). Global gene expression profiling has been undertaken to identify AM-regulated genes (Guimil et al., 2005, Hojnec et al., 2005). In many cases gene expression was monitored at the stage of a mature symbiosis and long lists of induced and suppressed genes have been created. The list of genes probably includes those essential for signaling during the initiation and formation of arbuscules (Paszkowski, 2006). Hence, formation of arbuscules within host cells is associated with dramatic morphological and physiological alterations in both the symbiotic partners. The molecular basis underlying arbuscule formation has elucidated the role of numerous genes that encode proteins, directly or indirectly involved in a signal transduction network that is required for the development of intracellular accommodation structures for symbiotic fungi by the host cell (Parniske, 2008).

#### The Symbiotic Interface and Nutrient Transfer

In root symbiosis, the symbiosome is the cellular environment hosting the microbial partner where the mutual exchange of nutrients and metabolites occurs. In AM, this is the cortical cell lumen harboring hyphal coils or arbuscules surrounded by the perihyphal or periarbuscular plasma membrane, respectively. The periarbuscular membrane, continuous with the plasma membrane of the cortical cell, is a key interface in the symbiosis; however, little is known of its lipid or protein composition or the mechanisms of its development (Pumplin and Harrison, 2009). Both the microbe- and the plant-derived symbiosome membranes tightly regulate the exchange of compounds, which is generally facilitated by membrane-integral transport proteins such as phosphate transport and the P-type H<sup>+</sup>-ATPase (Bucher, 2007). Recently, livecell imaging with fluorescently-tagged proteins has revealed the spatial and temporal information about the protein composition of the periarbuscular membrane. The study indicates that the periarbuscular membrane is composed of at least two distinct domains; an 'arbuscule branch domain' that contains the symbiosis-specific phosphate transporter, MtPT4, and an 'arbuscule trunk domain', that contains the blue copper-binding protein, MtBcp1 (Pumplin and Harrison, 2009).

Extensive transcript profiling has revealed numerous genes that are reported to be upregulated or repressed in mycorrhizae (Liu et al., 2003; Guimil et al., 2005; Hojnec et al., 2005; Kistner et al., 2005). Experimental evidence also exists for cell-specific localization of either transcripts or promoter activity of several genes involved in arbuscule function (i.e. P nutrition) and development. These include genes encoding P-type H<sup>+</sup>-ATPases (Krajinski et al., 2002) and P<sub>i</sub> transporter genes from tomato and potato (Nagy et al., 2005), Medicago truncatula (Harrison et al., 2002), Lotus japonicus (Maeda et al., 2006), and the cereals barley, wheat and maize (Glassop et al., 2005). Medicago truncatula serine carboxypeptidase (MtSCP1), a gene sharing identity with Ser carboxypeptidase II proteins from barley, wheat and Arabidopsis thaliana, and *MtCell*, a gene coding a membrane-anchored endo-1, 4- $\beta$ -D-glucanase-like protein, have been shown to be upregulated in the root cortex upon colonization of M. truncatula with Glomus versiforme (Liu et al., 2003). A mycorrhiza-specific class III chitinase gene is similarly regulated in cells containing developing or mature arbuscules (Bonanomi et al., 2001). The availability of expressed sequence tags (EST) of the model legume M. truncatula, and AM fungus, Glomus intraradices, permits identification of genes required for development of symbiotic interfaces. Candidate genes have been characterized and genes encoding one plant arabinogalactan protein (AGP) and three AGP-like (AGL) proteins have been identified. AGL proteins encoded by two AGL genes from G. intraradices (GiAGLs) represent a new class of AGPs not found in non-AM plants (Schultz and Harrison, 2008). Interestingly, several genes have been identified that are induced both in mycorrhized cortical cells and during rhizobial root colonization in legume species; for example, the early nodulin genes ENOD2, ENOD40 (van Rhijn et al., 1997) and ENOD11 (Chabaud et al., 2002), the Vicia faba leghemoglobin gene VFLb29 (Vieweg et al., 2004), and the gene calcium-binding protein 1 (Cbp1), which encodes a protein sharing similarities with calcium-binding proteins (Kistner et al., 2005). This supports the model outlined for the common symbiotic pathway in AM and rhizobiuminduced nodule development using signal transduction based on conserved mechanisms (Paszkowski, 2006; Bucher, 2007). Therefore, the transporters that mediate metabolite exchange at the interface between the plant and the fungus are of biotechnological interest, and some candidate genes encoding mycorrhizaspecific plant phosphate transporters have been cloned and sequenced (Parniske, 2008). These findings provide good support for the hypothesis that in infection processes in the symbioses between plant roots and two different microorganisms, i.e. AM fungi and rhizobial bacteria, similar signal perception and transduction cascades initiate mycorrhization and nodulation

(Parniske, 2004; Manchanda and Garg, 2007).

#### **3 Mycorrhizae and Their Diverse Roles**

Mycorrhizae are the rule in nature, not the exceptions. In this association the fungus takes over the role of the plant's root hair and acts as an extension of the root system (Muchovej, 2004). The beneficial effects of AM fungi result from one or several of the mechanisms. With mycorrhizal colonization in the roots, there is increased absorption surface area, greater soil area exposed, greater longevity of absorbing roots, better utilization of low-availability nutrients and better retention of soluble nutrients, thus reducing reaction with soil colloids or leaching losses (Muchovej, 2004; Selvaraj and Chelleppan, 2006). AM increase establishment, nodulation and atmospheric nitrogen fixation capacity in legumes (Turk et al., 2008). Mycorrhizae influence the colonization of roots by other microorganisms, and reduce the susceptibility of roots to soil-borne pathogens such as nematodes or phytopathogenic fungi (Selvaraj and Chelleppan, 2006). AM also modify soil-plant-water relations, thus promoting better adaptation of plants to adverse conditions, such as drought, salinity or heat stress (Fig. 2). At elevated heavy metal concentrations in soils, mycorrhizal fungi have been shown to detoxify the environment for plant growth (Muchovej, 2004). The real

significance of mycorrhizal fungi is that they connect the primary producers of ecosystems, plants, to the heterogeneously distributed nutrients required for their growth, enabling the flow of energy-rich compounds required for nutrient mobilization whilst simultaneously providing conduits for the translocation of mobilized products back to their hosts. Hence, understanding of the ecology and functioning of the AM symbiosis in the natural or agricultural ecosystem is essential for the improvement of plant growth and productivity.

## 3.1 Mycorrhizal Symbiosis and Mineral Nutrition

Obligately depending on plant photosynthates as energy sources, the extensive mycelial systems (the vegetative parts of the fungus) effectively explore soil substrates and acquire soil inorganic nutrients including the major macro-nutrients N, P and K and some micronutrients, Cu, Fe and Zn, with some capacity for acquiring organic nitrogen and phosphorus. These soilderived nutrients are not only essential for AM development but are also partly transferred to the host plant (Smith and Read, 1997; Leake et al., 2004).



Fig. 2 Functional diversity of arbuscular mycorrhizal (AM) symbiosis in terrestrial ecosystems: An overview

### 3.1.1 Phosphate Uptake Assisted by the AM Symbiosis

The major role of AM fungi is to supply infected plant roots with phosphorus, which is an extremely immobile element in soils (Bucher, 2007). Even if phosphorus is added to soil in soluble form it soon becomes immobilized as organic phosphorus, calcium phosphates or other fixed forms (Wetterauer and Killorn, 1996). AM fungi are known to be effective in increasing nutrient uptake, particularly phosphorus and biomass accumulation of many crops in low phosphorus soil (Osonubi et al., 1991).

### The Route of Symbiotic P<sub>i</sub> Uptake

AM fungi improve plant acquisition of phosphate (P<sub>i</sub>). It was recently shown that, depending on the particular plant-fungus combination, symbiotic phosphate uptake may partially participate or even dominate over all Pi acquisition (Smith et al., 2003). Strongly reduced mobility of P<sub>i</sub> in the soil and rapid P<sub>i</sub> uptake into the root lead to the development of a P<sub>i</sub> depletion zone around the root hair cylinder and a rapid decline in Pi acquisition over time (Marschner, 1995; Roose and Fowler, 2004). The extraradical mycelium of AM fungus grows far beyond the depletion zone, reaching a new pool of soluble phosphate (Smith and Read, 1997). Whereas in non-mycorrhizal roots the extension of the P<sub>i</sub> depletion zone is closely related to root hair length (Marschner and Dell, 1994), in mycorrhizal roots the depletion zone of P<sub>i</sub> greatly exceeds the root hair cylinder. This indicates that P<sub>i</sub>, which is not directly available to the plant, is delivered by the fungal hyphae. Thus, the presence of the P<sub>i</sub> depletion zone in the rhizosphere is a major factor contributing to the advantage of plants forming mycorrhizal associations. Strictly speaking, a mycorrhized plant does not constitute a rhizosphere but rather a 'mycorrhizosphere', composed of the rhizosphere and the hydrosphere. In this symbiotic system, the fungus bridges the mycorrhizosphere and P<sub>i</sub> is transported, in the form of polyphosphates, from the AM fungus soil interface to the intraradical symbiotic interface (Bucher, 2007).

The proposed metabolic route of symbiotic  $P_i$  acquisition starts with the assimilation of inorganic  $P_i$  at the hyphal-soil interface by fungal high-affinity transporters (Harrison and van Buuren, 1995; MaldonadoMendoza et al., 2001; Benedetto et al., 2005). Inside the fungus, inorganic P<sub>i</sub> is translocated in the form of polyphosphate from fungal structures outside of the root to those inside (Solaiman et al., 1999; Ohtomo and Saito, 2005). Before release into the periarbuscular interface, phosphate becomes depolymerized into inorganic P<sub>i</sub> (Ohtomo and Saito, 2005). P<sub>i</sub> is acquired from the interface by plant-encoded phosphate transporters. Such transporters have been identified from several plant backgrounds and were shown to be transcriptionally induced during the development of the AM symbiosis (Harrison et al., 2002; Paszkowski et al., 2002; Glassop et al., 2005; Nagy et al., 2005; Bucher, 2007; Javot et al., 2007; Pumplin and Harrison, 2009). Further, it has been demonstrated that the mycorrhizal Pi uptake pathway could dominate P<sub>i</sub> supply to plants irrespective of whether colonized plants exhibited improved growth and total P uptake (Smith et al., 2003, 2004). The fact that the AM fungal contribution to plant P<sub>i</sub> uptake is greater in a root-hairless mutant than its wild type indicates that fine tuning of both uptake pathways is required to meet the needs of the plants for this important nutrient (Jakobsen et al., 2005). To conclude, activation of the 'mycorrhizal' uptake pathway is therefore characterized by the induction of mycorrhiza-specific phosphate transporters and (partial) downregulation of the 'direct' uptake pathway phosphate transporters. Given that most phosphorus can be taken up via the 'mycorrhizal' uptake pathway, it can be hypothesized that mycorrhiza-upregulated plant phosphate transporters play a pivotal role in plant productivity and fitness in most natural and agricultural ecosystems.

#### Pht1 Genes Involved in the AM Symbiosis

Legumes such as *M. truncatula* and *L. japonicus* (Young et al., 2003) and solanaceous plants establish mutualistic AM symbiosis under natural conditions (Barker et al., 1998) and can thus be used as experimental systems for molecular-genetic work in mycorrhizae. Interestingly, a H<sup>+</sup>-ATPase gene exhibited arbuscule-specific expression in mycorrhizal tissue of *M. truncatula* (Krajinski et al., 2002), and a H<sup>+</sup>-ATPase protein was localized in the plant membrane around arbuscule hyphae in a tobacco mycorrhizal plant, which clarified the existence of nutrient transport activities at the interface between the two symbiotic organisms (Gianinazzi-Pearson et al., 2000). The identification of the potato P<sub>i</sub> transporter gene *StPT3*, which is expressed in cortical cells colonized by AM fungi, represented a starting point for a detailed analysis of P<sub>i</sub> transport at the AM symbiotic interface in solanaceous species. *StPT3* has been clearly identified as a high-affinity transporter. To date, one mycorrhizalinduced P<sub>i</sub> transporter coding gene has been identified from *M. truncatula* (*MtPT4*; Harrison et al., 2002; Javot et al., 2007), three have been found in solanaceous species (*StPT3/LePT3*; *StPT4/LePT4* and *StPT5/LePT5*; Karandashov and Bucher, 2005; Nagy et al., 2005), two in rice (*OsPT11* and *OsPT13*; Paszkowski et al., 2002; Guimil et al., 2005) and one in maize, barley and wheat (Glassop et al., 2005).

Despite the large number of mycorrhizal-inducible P<sub>i</sub> transporters identified to date, the functional genomics of P<sub>i</sub> transport at the symbiotic interface between AM fungi and host plants is not well understood. To date, the most intriguing study on the functional role of mycorrhizal-specific P<sub>i</sub> transporters originates from work on StPT3-like LjPT3 from Lotus japonicus (Maeda et al., 2006). Knockdown of the LjPT3 gene resulted in reduced growth of plants carrying transformed roots which were colonized by a mycorrhizal fungus, reduced allocation of radiotracer P<sub>i</sub> in the shoot, and decreased fungal colonization of mycorrhizae (Bucher, 2007). The identification of mycorrhiza - specific Pi transporters indicate the presence of a mycorrhiza-specific P<sub>i</sub> uptake system in vascular plants. Moreover, it is tempting to speculate that mycorrhizal P<sub>i</sub> transporters are involved in self/non-self recognition in mycorrhizae.

### 3.1.2 Nitrogen Transfer at the Mycorrhizal Interface

As well as benefiting plants by aiding phosphorus uptake from soil (Harrison and van Buuren, 1995), AM fungi can take up and transfer significant amounts of inorganic nitrogen ( $NH_4^+$  or  $NO_3^-$ ) to their host plants (He et al., 2003). The availability of nitrogen frequently limits plant growth, and depending on soil conditions, nitrogen transfer by mycorrhizal fungi can represent a significant route of uptake by the plant (He et al., 2003). AM fungi have been strongly implicated in the transfer of nitrogen from one plant to another (He et al., 2003), can increase the utilization of different forms of nitrogen by plants (Hodge et al., 2001) and have been shown to take up nitrogen directly and transfer it to host roots (He et al., 2003). Experimental observations have indicated that arginine is usually the principal nitrogenous product accumulated during periods of ammonium feeding at the uptake site, providing support for the importance of this amino acid in N transfer between fungal and plant cells.

The extraradical hyphae of AM fungi are able to take up and assimilate ammonium (NH<sub>4</sub><sup>+</sup>) (Johansen et al., 1992, 1993, 1996), nitrate (NO<sub>3</sub><sup>-</sup>) (Bago et al., 1996; Johansen et al., 1996) and amino acids (Hawkins et al., 2000; Hodge et al., 2001) from their surroundings and translocate N from diverse sources to the plant (Hawkins et al., 2000; Azcón et al., 2001; Vazquez et al., 2001). Assimilation of  $NH_4^+$  is a principal means of N absorption in AM fungal systems (Hawkins et al., 2000; Toussaint et al., 2004). N uptake and incorporation into amino acids via the glutamine synthetase/glutamate synthase (GS/GOGAT) cycle has been found in AM fungi (Smith et al., 1985). Stable isotope labeling has now suggested that inorganic nitrogen is taken up by the extraradical mycelium, incorporated into amino acids, translocated from extrato intraradical fungal structures as arginine and then transported as ammonium to the plant (Govindarajulu et al., 2005; Jin et al., 2005). Further support for this uptake route comes from the finding that transcript abundance of key enzymes of nitrogen assimilation and arginine breakdown preferentially accumulate in extra- and intraradical mycelia, respectively (Govindarajulu et al., 2005). A decline in the levels of the major amino acids present together with a decrease in the activity of fungal enzymes involved in the nitrogen assimilation during the mycorrhizal colonization process is seen (Blaudez et al., 1998). The extrusion of ammonia from fungal cells follows other pathways than those mediated by Amt proteins (ammonia transporter), either by passive efflux of the deprotonated form or by protein-mediated mechanisms. Thus, fungal cells are able to maintain a low cytoplasmic ammonia concentration, thus retaining a constant assimilatory capacity and in turn allow for sustained export into the plant root cells (Chalot et al., 2006). Therefore, the operation of a metabolic route assists the AM fungi in N uptake, transport and assimilation.

Various models have been proposed by Jin et al. (2005), Govindarajulu et al. (2005) and Chalot et al. (2006) which involve a number of steps for direct

transfer of ammonia from fungal to plant cells (i) AM fungi take up inorganic N (NH $_4^+$ , NO $_3^-$ ); (ii) absorbed N is mostly incorporated and stored in arginine; (iii) AM fungi assimilate the N through GS/GOGAT, asparagine synthase and the urea cycle; (iv) stored arginine can be co-transported with PolyP intact to the intraradical mycelium form the extraradical mycelium of AM fungi, and arginine is also bi-directionally transported within the extraradical mycelium; and (v) N released from transported arginine is transferred to the host as  $NH_4^+$  and can be incorporated into other free amino acids in mycorrhizal roots, while carbon (C) not transferred to the host is recycled back to the extraradical mycelium. In analogy to the path of symbiotic P<sub>i</sub> uptake, the arbuscule may be the site of symbiotic nitrogen uptake involving plant encoded nitrogen transporters located within the periarbuscular plant membrane (Paszkowski, 2006). Apart from the role of AM fungi in facilitating the uptake of inorganic N by plants, AM fungi also give their host plant the ability to use organic matter as a source of N (Hodge et al., 2001). A key role of the AM symbiosis in linking the process of N mineralization to plant N demand in soil, where the AM symbiosis regulates the recycling of plant residue into living plant biomass, thus impacting the structure of the soil microbial community, has been verified (Leigh et al., 2009; Nayyar et al., 2008). The mechanisms involved in the fungal delivery of N are a matter of considerable interest because, depending on N availability and mobility and given the near-ubiquity of the AM symbiosis, these processes may represent a significant nutritional benefit to the plant. The existence of this pathway and the high flux of nitrogen through it indicate that the arbuscular mycorrhizal symbiosis can effectively transfer large amounts of nitrogen from the soil to plant roots (Jackson et al., 2008).

## 3.2 Alleviation of Salt Stress by Arbuscular Mycorrhizal (AM) Fungi

Soil salinization is an ever-increasing threat to the cultivation of crop plants around the globe, with this problem being serious in arid and semi-arid areas. The arbuscular mycorrhizal (AM) symbiosis is a mutually beneficial interaction and is an important integral component of the natural ecosystem. AM fungi occur naturally in saline environments (Garcia and Mendoza, 2007) and have been shown to increase plant yield in saline soils. In saline and sodic soils, drainage is poor and salt accumulates on the surface of the soil, thus adversely affecting plant growth. Salinity affects the formation and function of mycorrhizal symbiosis (Giri et al., 2003; Juniper and Abbott, 2006). However, several studies have demonstrated that inoculation with mycorrhizal fungi improves growth and productivity of plants under a variety of salt stress conditions (Giri and Mukerji, 2004; Giri et al., 2007). Recently, many researchers reported that AM fungi could enhance the ability of plants to cope with salt stress (Yano-Melo et al., 2003; Rabie, 2005; Cho et al., 2006; Ghazi and Al-Karaki, 2006; Sannazzaro et al., 2006) by improving uptake of plant nutrients such as P, N, Zn, Cu and Fe (Cantrell and Lindermann, 2001; Asghari et al., 2005; Ghazi and Al-Karaki, 2006). The improvement in the plant P status has been suggested as the most important strategy of salinity stress tolerance in AM colonized plants (Giri et al., 2003). However, other studies have shown that salt stress tolerance of AM plants is not always related to improved P status (Feng et al., 2002), and other physiological processes help in AM plant growth improvement under such conditions.

AM fungi can play an important role in increasing the carbon dioxide exchange rate, transpiration and stomatal conductance (Feng et al., 2002), improving ion imbalance (Zandavalli et al., 2004; Giri et al., 2007), protecting enzyme activity (Giri and Mukerji, 2004; Rabie and Almadini, 2005), facilitating water uptake (Feng et al., 2002; Sheng et al., 2008; Colla et al., 2008), and favorably adjusting the osmotic balance and composition of carbohydrates (Ruiz-Lozano, 2003).

Thus, these microorganisms play an important role in soil productivity and plant nutrition under a variety of salt stress conditions. Owing to the importance of AM fungi under salt stress conditions, they have been considered as bio-ameliorators of saline soils. Mycorrhizal colonization can improve the physiological performance of stressed plants, leading to higher yields and quality. The induction of resistance is provided through a discriminated absorption of the ions present in the circulating solution, through better balance of mineral nutrient uptake, by influencing the hormonal balance of the host plant or by increasing water uptake.

### 3.3 Plant Water Relationship

Although most of the work done with AM fungi has concentrated on their effects in plant mineral nutrition, there is also increasing interest in drought resistance of mycorrhizal plants (Allen and Boosalis, 1983). AM fungi are important in sustainable agriculture because they improve plant water relations and thus increase the drought resistance of host plants (Allen and Allen, 1986; Nelsen, 1987). Improved plant water status and changes in water relations have been attributed to a wide variety of mechanisms, including some mechanisms not directly related to phosphorus nutrition or water uptake (Davies et al., 1992). The abilities of specific fungus-plant associations to tolerate drought are of great interest (Ruiz-Lozano et al., 1995). AM fungi infection has been reported to increase nutrient uptake in water-stressed plants (Busse and Ellis, 1985), enabling plants to use water more efficiently and to increase root hydraulic conductivity (Graham and Syversen, 1984). Root water uptake depends on root hydraulic conductance, which is ultimately governed by aquaporins (Luu and Maurel, 2005). Aquaporins are membrane intrinsic proteins that form a pore in all cell membranes of living organisms, facilitating the passive water flow through membranes following an osmotic gradient (Kruse et al., 2006). Plasma membrane intrinsic proteins (PIPs) regulate all water transport through plant tissues. The plants over-expressing or lacking one or more PIP genes have more or less root water uptake capacity, respectively (Aharon et al., 2003; Javot et al., 2003). In fact, AM plants are frequently more tolerant to drought and salt stresses than non-AM plants (Rosendahl and Rosendahl, 1991; Khalvati et al., 2005; Al-Karaki, 2006; Porcel et al., 2006; Aroca et al., 2007). AM plants are able to take up more water from the soil than non-AM plants under water-deficient conditions (Marulanda et al., 2003; Khalvati et al., 2005). However, this capacity depends on the fungal species, Glomus intraradices being one of the most efficient AM fungi in enhancing plant water uptake from the soil among six fungi tested (Marulanda et al., 2003).

Aroca et al. (2007) evaluated how AM symbiosis influences root hydraulic properties, aquaporin expression and abundance in roots of *Phaseolus vulgaris* L. plants under drought, cold or salinity conditions. They reported that colonization of *P. vulgaris* roots by the AM fungus *Glomus intraradices* prevented leaf dehydration caused by drought and salinity treatments as revealed by the higher relative water content (RWC) of AM leaves compared with non-AM leaves. These results confirm the beneficial effect of AM fungi in host plant water status under these two stresses (Rosendahl and Rosendahl, 1991; Porcel et al., 2006). The protection of mycorrhizal plants against water stress is related to the effects that the endophytes have on increasing leaf conductance and transpiration as well as P and K uptake. Potassium plays a key role in plant water stress and has been found to be the cationic solute which is responsible for stomatal movement in response to changes in bulk leaf water status (Ruiz-Lozano et al., 1995). Thus, AM symbiosis regulates root hydraulic properties and enhances root hydraulic conductance tolerance to drought, cold and salinity stresses (Aroca et al., 2007). Amelioration of drought stress by different AM fungal species can be ascribed to specific physiological (CO2 fixation, transpiration, water-use efficiency) and nutritional (P and K) mechanisms according to the fungus involved in the symbiotic association. Suitably adapted AM fungal isolates are potentially important for maintaining and restoring the plant-soil equilibrium in sustainable agriculture situations (Ruiz-Lozano et al., 1995). The role played by AM fungi in alleviating water stress of plants has been investigated and it appears that drought resistance is enhanced. The precise mechanisms underpinning this are still in doubt but it could be an indirect effect of the extraradical mycelium improving nutrient absorption. The alleviation of water stress is not only limited to arid or semi-arid zones of the planet, but also where short-term droughts occur; an increased reliance on AM fungi for nutrient uptake can be frequently detected.

## 3.4 Protection of Host Roots from Pathogens

Multitrophic interactions are powerful forces shaping the structure of living communities. Plants encounter a great diversity of organisms in their environment: some of these interactions are beneficial, e.g. symbiotic fungi and insect pollinators, while some are detrimental, e.g. herbivore insects and pathogenic microorganisms. These interactions between belowground and above-ground organisms are receiving increasing attention because they may influence plant defenses against biotic and abiotic stresses (van Dam et al., 2000). AM symbiosis is mutualistic interaction between plant roots and soil fungi that is considered beneficial to the plant because of the increased uptake of phosphorus and other scarcely mobile nutrients by mycorrhizal plants. Several investigations have demonstrated their positive impact on nutrient uptake and growth of banana (Jaizme-Vega and Azcón, 1995; Yano-Melo et al., 1999), improved resistance to abiotic stresses (Rufyikiri et al., 2000; Ruiz-Lozano, 2003) and biotic stresses caused by nematodes (Jaizme-Vega et al., 1997; Elsen et al., 2001) and *Fusarium oxysporum* f. sp *cubense*, the causal agent of panama disease (Jaizme-Vega et al., 1998).

Most studies on the interactions between AM fungi and plant parasitic nematodes reported that root colonization by AM fungi increases tolerance of the host to Meloidogyne species, such as that of tomato and white cover to Meloidogyne hapla (Cooper and Grandison, 1986); peanut to M. arenaria (Carling et al., 1996); banana to M. incognita (Jazme et al., 1997) and Prunus root stocks to M. javanica (Calvet et al., 2001). Colonization by AM fungi induces resistance or tolerance to a variety of pathogens in tomato and in other plants (Trotta et al., 1996; Lingua et al., 2002). These changes are mediated by a variety of mechanisms, including the upregulation and downregulation of specific genes (Tahiri-Alaoui and Antoniw, 1996), that result in localized and systemic responses by the plant. These responses include the synthesis of new isoforms of chitinases and glucanases and the thickening of the cell walls (Azcón-Aguilar et al., 2002; Pozo et al., 2002), that may affect herbivore colonization. Declerck et al. (2002) carried out investigations on the interaction between four AM fungi, Glomus sp., Glomus proliferum, G. intraradices and G. versiforme, and the root-rot fungus Cylindrocladium spathiphylli. Pre-inoculation of plants with AM fungi attenuated the detrimental effect of the pathogen. These results corroborate previous studies involving other plants grown in the presence of root pathogenic fungi, as these plants were less affected when colonized with AM fungi (Jaizme-Vega et al., 1998; Abdalla and Abdel-Fattah, 2000). The development of root-rot symptoms on bananas was considerably affected by the presence of AM fungi. The lower disease severity observed in mycorrhizal banana plants was associated with improved growth of the plants. Therefore, AM fungal symbiosis mainly affected the host-pathogen relationship by improving P nutrition, leading to greater resistance to root-infecting fungi. More recently, the effects of different species of AM fungi on parasitism rates have been reported (Gange et al., 2003) but this study did not demonstrate a direct link between AM and attraction of insect parasitoids.

Guerrieri et al. (2004) tested the hypothesis that an AM symbiosis makes tomato plants significantly more attractive towards the aphid parasitoid, Aphidius ervi (Haliday), which is well known for its efficiency against the potato aphid pest. The positive effect of mycorrhizal colonization on the attraction of the parasitic wasp, A. ervi, to its host plants and a negative effect on insect pest populations observed in this study have important implications for understanding of insect population dynamics and predicting the plant defense mechanisms. Castillo et al. (2006) investigated the effects of single and joint inoculation of olive planting stocks cvs. Arbequina and Picual with the AM fungi, G. intraradices, G. mosseae or G. viscosum, and the root knot nematodes M. incognita and M. javanica on plant performance and nematode infection. They concluded that prior inoculation of olive plants with AM fungi contributed to improved health status and vigor and reduced the severity of root galling by root-knot nematode. In conclusion, these results demonstrate the positive impact of AM fungi on P nutrition and plant growth under root pathogen pressure. However, this does not imply that better P nutrition alone accounts for increased resistance against phytopathogens. In the past decade, some authors have suggested that other mechanisms could be involved: proposed mechanisms include activation of the plant defense system and disease resistance (Benhamou et al., 1994; St. Arnaud et al., 1994); direct or indirect competition in the rhizosphere (St. Arnaud et al., 1994); biochemical changes in the plant and anatomical changes in the roots (Benhamou et al., 1994; Hooker et al., 1994); and competition for host resources in root tissues (Hooker et al., 1994). Hence, AM fungi have been shown to increase resistance to root-infecting pathogenic fungi and root-invading nematodes. The reasons for increased resistance to pathogens are not well understood, but a well-established AM infection is a prerequisite for protection. This is an intriguing area of research and bioprotection may be the primary role for AM fungi in some natural ecosystems rather than nutrient acquisition. The current emphasis on low

input-based agrotechnology for crop production systems has stimulated the use of AM fungi as bioprotectors, phytostimulators or biofertilizers against plant diseases caused by soil-borne pathogens.

### 3.5 Mycorrhizoremediation

Biosphere pollution by heavy metals and nucleotides has been accelerated dramatically during the last few decades due to mining, smelting, manufacturing, treatment of agricultural soils with agro-chemicals and soil sludge, etc. (He et al., 2005). Heavy metals such as lead (Pb), arsenic (As), cadmium (Cd) and mercury (Hg), being added to our soils through industrial, agricultural and domestic effluents, persist in soils and can either be adsorbed in soil particles or leached into groundwater. Phytoremediation (such as phytoextraction, phytostabilization and rhizofiltration) of soils contaminated by heavy metals has been widely accepted as a cost-effective and environmentally friendly clean-up technology. However, the progress in this field is hindered by lack of understanding of complex interactions in the rhizosphere and plant-based mechanisms which allow metal translocation and accumulation in plants (Yu et al., 2004). Complex interactions between roots, microorganisms and fauna in the rhizosphere have a fundamental effect on metal uptake and plant growth. Some AM fungi are adapted to adverse conditions so they can benefit plants under a variety of environmental stresses. AM fungi are involved in plant interactions with soil toxic metals, either by alleviating metal toxicity to the host or by accentuating it (Meharg, 2003; Pawlowska and Charvat, 2002). Despite the significant role that AM fungi play in plant interactions with soil toxic metals and the ubiquity of AM fungi in the soil environment, only recently has progress been made towards understanding the cellular mechanisms utilized by AM fungi to metabolize heavy metals and alleviate their cytotoxicity (Lanfranco et al., 2002). Numerous studies have revealed that AM fungi confer upon plants tolerance against heavy metal stress (Davies et al., 2001; Levyal et al., 2002; Liao et al., 2003). Plant associations with AM fungi are suggested as a potential biological solution to ameliorate plant resistance to metal toxicity and restore fertility of soils polluted by heavy metals such as Cd (Vivas et al., 2005). A well-developed mycorrhizal symbiosis may enhance the survival of plants in polluted areas by better nutrient acquisition (P, N, Zn, Cu, Fe, etc.), water relations, pathogenic resistance, phytohormone production, contribution to soil aggregation, amelioration of soil structure, and thus improved success of all kinds of bioremediation (Jeffries et al., 2003; Gaur and Adholeya, 2004). Immobilization of metals in the fungal biomass is proposed as a mechanism by which these fungi may increase plant tolerance to heavy metals. Mycorrhizal roots may act as a barrier against metal transport, reducing transfer and enhancing root/shoot Cd ratios (Andrade and Silveria, 2008). This effect is attributed to metal adsorption on hyphal walls, since chitin has an important metal-binding capacity (Joner et al., 2000; Bi et al., 2003; Christie et al., 2004). Recently, it has been suggested that glomalin, a glycoprotein produced by AM fungi, may have a metal chelating function, diminishing metal availability for plants (Zhu, 2001; Gonzalez-Chavez et al., 2004; Khan et al., 2006; Saleh and Saleh, 2006). Another possible mechanism of metal tolerance includes dilution of metal concentration in plant tissues due to the promotion of plant growth by AM fungi (Andrade and Silveria, 2008), uptake exclusion by precipitation or chelation in the rhizosphere (Kaldorf et al., 1999), and P-mediated effects on the host plant (Wang et al., 2005). The AM fungi associated with metal-tolerant plants may contribute to the accumulation of heavy metals in roots in a non-toxic form inside hyphal cell walls or complexed into phosphate materials inside the cells (Galli et al., 1995). However, the effect of AM fungi on the uptake of metals by plants is not yet totally clear, with reports finding increased uptake of metals (Tonin et al., 2001; Liao et al., 2003; Whitefield et al., 2004; Citterio et al., 2005), decreased uptake (Weissenhorn et al., 1995, Chen et al., 2003) or no uptake effects (Trotta et al., 2006; Wu et al., 2001). Factors such as the metal element and its availability, plant species, fungal species or strain and differences in mycorrhizal and non-mycorrhizal plant size and P content in experiments may account for different results (Levyal and Joner, 2000).

AM fungi, therefore, play a significant ecological role in the phytostabilization of potentially toxic trace element-polluted soils by sequestration and, in turn, help mycorrhizal plants to survive in polluted soils. They can alter plant productivity, because AM fungi can act as biofertilizers, bioprotectants or biodegraders (Xavier and Boyetchko, 2002). Their potential role in phytoremediation of heavy metal-contaminated soils and water is also becoming evident (Khan, 2001; Jamal et al., 2002; Hayes et al., 2003). Thus, the benefits of mycorrhizae may be associated with metal tolerance and also with metal plant nutrition. Therefore, in degraded and contaminated soils, that are often poor in nutrients and with low water-holding capacities, mycorrhizae formation would be of great importance. In conclusion, a biotechnological goal is to use a combined inoculation of selected rhizosphere microorganisms to minimize toxic effects of pollutants and to maximize plant growth and nutrition (Saleh and Saleh, 2006). The ecological complexity and diversity of plant-microbe-soil combinations and the role that AM play in phytorestoration of heavy metal-contaminated soils, i.e. mycorrhizoremediation, thus need to be taken into consideration to restore the fertility of heavy metal soil (Khan et al., 2006). Ecosystem restoration of heavy metal-contaminated soil practices need to incorporate microbial biotechnology research and development. Since AM fungi are reported to be present on the roots of plants growing on heavy metal-contaminated soils, isolation of indigenous stress-adapted AM fungi can be a potential biotechnological tool for inoculation of plants in disturbed ecosystems.

## 3.6 Miscellaneous Roles of Arbuscular Mycorrhizal (AM) Fungi

Roots colonized by AM fungi are often thicker and carry fewer root hairs. Such changes in morphology are expected to be under phytohormonal control (Selvaraj, 1998). Abscisic acid (ABA) was found to be considerably enhanced in both roots and shoots of AM plants as compared with non-mycorrhizal control (Danneberg et al., 1992). Also, an increase in Indole Acetic Acid (IAA), gibberellin and cytokinin level was observed in *G. fasciculatum*-inoculated *P. juliflora* recorded by Selvaraj (1998), showed the influence of the AM fungi, *G. fasciculatum*, on increased level of growth hormones. Barea and Azcon-Aguilar (1982) also noted that in axenic experiments, mycorrhizal fungi produced auxin-, gibberellin- and cytokinin-like substances and stimulated plant growth.

The mycorrhizal plants also showed increased production of certain enzymes. Increased peroxidase is one of the most widespread biochemical activities in diseased and injured plant tissues. Spanu and Bonfante-Fasolo (1998) measured the cell wall-bound peroxidase in Allium porrum during root growth and development of G. versiforme. Pacovsky et al. (1990) studied peroxidase activity in Phaseolus vulgaris infected by G. etunicatum and found that peroxidase activity increased in the mycorrhizal plants. Phosphatases of the mycorrhizae are both specifically induced in the presence of Glomus spores and are sensitive to the level of phosphate in the environment (Pacovsky et al., 1991). Selvaraj (1998) found that due to inoculation of AM fungi, G. fasciculatum acid phosphatase activity was increased in leaves and roots of P. juliflora. Tisserant et al. (1993) observed that histochemical tests revealed the presence of alkaline phosphatase in Glomus-infected roots of A. porrum and Platamus acerifolia. The presence of AM specific alkaline phosphatase activity in A. cepa and P. accindentalis plants inoculated with G. mosseae has been reported (Gianinazzi-Pearson and Gianinazzi, 1976). Mycorrhizae also enable plants to survive in disturbed and endangered ecosystems, as those are polluted by radioactive elements. In an atomic power station, products of nuclear fission reactions such as cesium, <sup>137</sup>Cs, and strontium, <sup>90</sup>Sr, are regularly released into the environment as a result of weapons testing, nuclear power production and nuclear fuel reprocessing (Haury and Schikarski, 1997). Plants absorb the Cs and Sr less efficiently than their nutrient analogues potassium and calcium, respectively (Zhu and Smolders, 2000). Both Cs and Sr are taken up from soil solution by plants as  $K^+$  and  $Ca^{2+}$ , as they are similar to those cations in chemical properties (White and Bradley, 2000). AM fungi are beneficial for the uptake of nutrients and also for plants to survive, even from a disturbed soil after radionuclide deposition (Ahiabor and Hirata, 1994). Selvaraj et al. (2004) studied the effect of AM spores, G. fasciculatum, on two plants, Phyllanthus niruri and Ecliptica alba, and revealed that AM fungi was more beneficial for plant growth. It was noticed that in spite of growth disturbance, AM fungi-inoculated plants were efficient in tolerating the endangered ecosystems. The two experimental host plants showed more uptake of K<sup>+</sup> and Ca<sup>2+</sup> in the roots of AM-inoculated plants as compared with the uninoculated plants. Thus, through the help of AM fungi and the soil's nature to hold the radionuclide to prevent the expression of radioactivity, chances are greater for the vegetation to survive in the

disturbed ecosystems in a better way. There is very strong circumstantial evidence therefore that AM fungi would enhance uptake and recycling of radionuclides, particularly <sup>137</sup>Cs and strontium, <sup>90</sup>Sr (Selvaraj et al., 2004). Thus, plants receive support from AM fungi, with the help of its symbiotic association, in the aspect of uptake of phosphorus and other nutrients, enhancement of growth hormones, increase in protein content, increase tolerance to drought and heavy metals, increase in salinity tolerance and resistance to root-borne pathogens. Also, AM fungi can help in uptake of radionuclides in a disturbed ecosystem with radioactive elements (Selvaraj and Chelleppan, 2006).

So, not only do the activities of AM fungi have multiple functions that enhance plant performance and productivity, but they also play crucial roles in the development of soil properties and the health of the entire ecosystem. Thus, with important nutrient uptake functions of AM fungi being appropriately managed, mycorrhizae can potentially offer a more effective and sustainable element biofortification to curb human malnutrition.

### **4** Conclusion

The increasing consumer demands for organic or sustainably-produced food requires the incorporation of natural roles of microorganisms, such as arbuscular mycorrhizal (AM) fungi. It is clearly revealed from the preceding discussion that mycorrhizal symbiosis plays fundamental roles in shaping terrestrial ecosystems and must be considered as an essential factor for soil fertility and for promoting plant health and productivity. Over the past few years, a novel and unexpected developmental capacity of plant cells has been discovered that is essential for the intracellular uptake of AM fungi. Recent progress in the molecular and genetic analysis of mycorrhizal development has provided novel insights into the evolution of this symbiosis. Elucidation of molecular events associated with signaling and nutrient acquisition processes has moved rapidly forward. Several genes (GmGin1, Lj POLLUX, Lj Nup 133, Lj SYM 24, AM-specific Pht1-encoding genes and plant encoded nitrogen transporters) and signaling cascades have been identified which play a role in the establishment of mycorrhizal symbiosis. Collectively, these fungal symbionts confer tolerance to drought,

heat, salinity, osmotic stress, disease and herbivory, and thus promote growth and nutrient acquisition. AM symbiosis occurs in almost all habitats, including disturbed soils contaminated with heavy metals, and plays an important role in metal tolerance. Thus, we can conclude that some plants are unable to endure habitat-imposed abiotic and biotic stresses in the absence of fungal endophytes. The use of stress-tolerant mycorrhizal fungi may be a promising strategy to develop tools for soil reclamation and amelioration.

Despite the identification of genes, proteins and enzymes involved in fungal and plant perception, and P and N transport and assimilation, very little is known about how these elements are transferred to the hosts. Thus, the understanding of molecular mechanisms and signaling pathways coupled to AM symbiosis needs further refinement and there is a need to unravel the complexity of the biology of AM symbiosis.

### References

- Abd-Alla M.H., Omar S.A., Karanxha S. (2000) The impact of pesticides on arbuscular-mycorrhizal and nitrogen-fixing symbioses in legumes, Appl. Soil Ecol. 14, 191–200.
- Abdalla M.E., Abdel-Fattah G.M. (2000) Influence of the endomycorrhizal fungus *Glomus mosseae* on the development of peanut pod rot disease in Egypt, Mycorrhiza 10, 29–35.
- Aharon R., Shahak Y., Wininger S., Bendov R., Kapulnik Y., Galili G. (2003) Overexpression of a plasma membrane aquaporin in transgenic tobacco improves plant vigor under favorable growth conditions but not under drought or salt stress, Plant Cell 15, 439–447.
- Ahiabor B.D., Hirata H. (1994) Characteristic responses of three tropical legumes to the inoculation of two species of VAM fungi in Andosol soils with different fertilities, Mycorrhiza 5, 63–70.
- Akiyama K., Matsuoka H., Hayashi H. (2002) Isolation and identification of a phosphate deficiency-induced C-glycosylflavonoid that stimulates arbuscular mycorrhiza formation in melon roots, Mol. Plant-Microbe In. 15, 334–340.
- Akiyama K., Matsuzaki K., Hayashi H. (2005) Plant sesquiterpenes induce hyphal branching in arbuscular mycorrhizal fungi, Nature 435, 824–827.
- Al-Karaki G.N. (2006) Nursery inoculation of tomato with arbuscular mycorrhizal fungi and subsequent performance under irrigation with saline water, Sci. Hortic. 109, 1–7.
- Allen E.B., Allen M.F. (1986) Water relations of xeric grasses in the field: interactions of mycorrhizas and competition, New Phytol. 104, 559–571.
- Allen M.F., Boosalis M.G. (1983) Effects of two VAmycorrhizal fungi on drought tolerance of winter wheat, New Phytol. 93, 67–76.

- Andrade, S.A.L., Silveria, A.P.D. (2008) Mycorrhiza influence on maize development under Cd stress and P supply, Braz. J. Plant Physiol. 20, 39–50.
- Aroca R., Porcel R., Ruiz-Lozano J.M. (2007) How does arbuscular mycorrhizal symbiosis regulate root hydraulic properties and plasma membrane aquaporins in *Phaseolus vulgaris* under drought, cold or salinity stresses? New Phytol. 173, 808–816.
- Aryal U.K., Xu H.L. (2001) Mycorrhizal associations and their manipulation for long-term agricultural stability and productivity, J. Crop Prod. 3, 285–302.
- Asghari H., Marschner P., Smith S., Smith F (2005) Growth response of *Atrilpex nummularia* to inoculation with arbuscular mycorrhizal fungi at different salinity levels, Plant Soil 273, 245–256.
- Azcón R., Ruiz-Lozano J.M., Rodriguez R. (2001) Differential contribution of arbuscular mycorrhizal fungi to plant nitrate uptake of <sup>15</sup>N under increasing N supply to the soil, Can. J. Bot. 79, 1175–1180.
- Azcón-Aguilar C., Jaizme-Vega M.C., Calvet C. (2002) Plant defense responses induced by arbuscular mycorrhizal fungi, in: Gianinazzi S., Schuepp H., Barea J.M. (Eds.), Mycorrhizal Technology in Agriculture, Birkhauser press, Basel, pp. 187–198.
- Bago B., Vierheiling H., Piche Y., Azcon-Aguilar C. (1996) Nitrate depletion and pH changes by the extraradical mycelium of the arbuscular mycorrhizal fungus *Glomus mosseae* grown in monoxenic culture, New Phytol. 133, 273–280.
- Barea J.M., Azcon-Aguilar C. (1982) Production of plant growth regulating substances of VAM fungus *Glomus mosseae*, Appl. Environ. Microb. 43, 810–813.
- Barker S., Stummer B., Gao L., Dispain I., O'Connor P., Smith S. (1998) A mutant in *Lycopersicon esculentum* Mill. with highly reduced VA mycorrhizal colonization: isolation and preliminary characterization, Plant J. 15, 791–797.
- Bécard G., Pfeffer P.E. (1993) Status of nuclear division in arbuscular mycorrhizal fungi during in vitro development, Protoplasma 174, 62–68.
- Bécard G., Kosuta S., Tamasloukht M., Séjalon-Delmas N., Roux C. (2004) Partner communication in the arbuscular mycorrhizal interaction, Can. J. Bot. 82, 1186–1197.
- Bécard G., Taylor L.P., Douds D.D. Jr., Pfeffer P.E., Doner L.W. (1995) Flavonoids are not necessary plant signal compounds in arbuscular mycorrhizal symbioses, Mol. Plant-Microbe In. 8, 252–258.
- Benedetto A., Magurno F., Bonfante P., Lanfranco, L. (2005) Expression profiles of a phosphate transporter gene (*GmosPT*) from the endomycorrhizal fungus *Glomus mosseae*, Mycorrhiza 15, 620–627.
- Benhamou N., Fortin J.A., Hamel C., St Arnaud M., Shatilla A. (1994) Resistance response of mycorrhizal Ri T-DNAtransformed carrot roots to infection by *Fusarium oxysporum* f.sp. chrysanthemi, Phytopathol. 84, 958–968.
- Bethlenfalvay G.J., Schüepp H. (1994). Arbuscular mycorrhizae and agrosystem stability, in: Gianinazzi S., Schüepp H. (Eds.), Impact of Arbuscular Mycorrhizas on Sustainable Agriculture and Natural Ecosystem, Birkhäuser Verlag Basal, Switzerland, pp. 117–131.
- Bi Y.L., Li X.L., Christie P. (2003) Influence of early stages of arbuscular mycorrhiza on uptake of zinc and phosphorus by

red clover from low-phosphorus soil amended with zinc and phosphorus, Chemosphere 50, 831–837.

- Blaudez D., Chalot M., Dizengremel P., Botton B. (1998) Structure and function of the ectomycorrhizal association between *Paxillus involutus* (Batsch) Fr. and *Betula pendula* (Roth.). II. Metabolic changes during mycorrhiza formation, New Phytol. 138, 543–552.
- Blee K.A., Anderson A.J. (1998) Regulation of arbuscule formation by carbon in the plant, Plant J. 16, 523–530.
- Bonanomi A., Wiemken A., Boller T., Salzer P. (2001) Local induction of a mycorrhiza-specific class III chitinase gene in cortical root cells of *Medicago truncatula* containing developing or mature arbuscules, Plant Biology 3, 194–199.
- Bonfante P., Bianciotto V. (1995) Presymbiontic versus symbiontic phase in arbuscular endomycorrhizal fungi: morphology and cytology, in: Varma A., Hock B. (Eds.), Mycorrhiza, Springer Press, Berlin, pp. 229–247.
- Bonfante P., Genre A. (2008) Plants and arbuscular mycorrhizal fungi: an evolutionary developmental perspective, Trends Plant Sci. 13, 492–498
- Bonfante-Fasolo P., Perotto S. (1992) Plant and endomycorrhizal fungi: the cellular and molecular basis of their interaction, in: Verma D. (Ed.), Molecular signals in plant-microbe communications, CRS Press, Boca Raton, FL, USA, pp. 445–470.
- Bouwmeester H.J., Matusova R., Zhongkui S., Beale, M.H. (2003) Secondary metabolite signalling in host–parasitic plant interactions, Curr. Opin. Plant Biol. 6, 358–364.
- Breuninger M., Requena N. (2004) Recognition events in AM symbiosis: analysis of fungal gene expression at the early appressorium stage, Fungal Genet. Biol. 41, 794–804.
- Bucher M. (2007) Functional biology of plant phosphate uptake at root and mycorhiza interfaces, New Phytol. 173, 11–26.
- Buee M., Rossignol M., Jauneau A., Ranjeva R., Becard G. (2000) The pre-symbiotic growth of arbuscular mycorrhizal fungi is induced by a branching factor partially purified from plant root exudates, Mol. Plant-Microbe In. 13, 693–698.
- Busse M.D., Ellis J.R. (1985) Vesicular-arbuscular mycorrhizal (*Glomus fasiculatum*) influence on soybean drought tolerance in high phosphorus soil, Can. J. Bot. 63, 2290–2294.
- Calvet C., Pinochet J., Hernández Dorrego A., Estaún V., Camprubí A. (2001) Field microplot performance of the peach–almond hybrid GF-677 after inoculation with arbuscular mycorrhizal fungi in a replant soil infested with rootknot nematodes, Mycorrhiza 10, 295–300.
- Cantrell I.C., Lindermann R.G. (2001) Preinoculation of lettuce and onion with VA mycorrhizal fungi reduces deleterious effects of soil salinity, Plant Soil 233, 269–281.
- Carling D.E., Roncadori R.W., Hussey R.S. (1996) Interactions of arbuscular mycorrhizae, *Meloidogyne arenaria*, and phosphorus fertilization on peanut, Mycorrhiza 6, 9–13.
- Castillo P., Nico A.I., Azcón-Aguilar C., Del Río Rincón C., Calvet C., Jiménez-Díaz R.M. (2006) Protection of olive planting stocks against parasitism of root-knot nematodes by arbuscular mycorrhizal fungi, Plant Pathol. 55, 705–713.
- Chabaud M., Venard C., Defaux-Petras A., Becard G., Barker D.G. (2002) Targeted inoculation of *Medicago truncatula in vitro* root cultures reveals *MtENOD11* expression during early stages of infection by arbuscular mycorrhizal fungi, New Phytol. 156, 265–273.

- Chalot M., Blaudez D., Annick B. (2006) Ammonia: a candidate for nitrogen transfer at the mycorrhizal interface, Trends Plant Sci. 11, 263–266.
- Chen B.D., Li X.L., Tao H.Q., Christie P., Wong M.H. (2003) The role of arbuscular mycorrhiza in zinc uptake by red clover growing in a calcareous soil spiked with various quantities of zinc, Chemosphere 50, 839–846.
- Cho K., Toler H., Lee J., Ownley B., Stutz J.C., Moore J.L. (2006) Mycorrhizal symbiosis and response of sorghum plants to combined drought and salinity stresses, J. Plant Physiol. 163, 517–528.
- Christie P., Li X., Chen B. (2004) Arbuscular mycorrhiza can depress translocation of zinc to shoots of host plants in soils moderately polluted with zinc, Plant Soil 261, 209–217.
- Citterio S., Prato N., Fumagalli P., Aina R., Massa N., Santagostino A. (2005) The arbuscular mycorrhizal fungus *Glomus mosseae* induces growth and metal accumulation changes in *Cannabis sativa* L., Chemosphere 59, 21–29.
- Colpaert J.V. (2008) Heavy metal pollution and genetic adaptations in ectomycorrhizal fungi, in: Avery S., Stratford M., van West P. (Eds.), Stress in yeasts and filamentous fungi, Elsevier, Amsterdam, pp. 157–173.
- Colla G., Rouphael Y., Cardarelli M., Tullio M., Rivera C.M., Rea E. (2008) Alleviation of salt stress by arbuscular mycorrhiza in zucchini plants grown at low and high phosphorus concentration, Biol. Fert. Soils 44, 501–509.
- Cooper K.M., Grandison G.S. (1986) Interaction of vesicular– arbuscular mycorrhizal fungi and root-knot nematode on cultivars of tomato and white clover susceptible to *Meloidogyne hapla*, Ann. Appl. Biol. 108, 555–566.
- Danneberg G., Latus C., Zimmer W., Hundeshagen B., Schneider-Poetsch Hj., Bothe H. (1992) Influence of vesicular arbuscular mycorrhiza on phytohormone balances in maize (*Zea mays* L.), J. Plant Physiol. 141, 33–39.
- Davies F.T., Potter J.R., Linderman R.G. (1992) Mycorrhiza and repeated drought exposure affect drought resistance and extraradical hyphae development of pepper plants independent of plant size and nutrient content, J. Plant Physiol. 139, 289–294.
- Davies F.T., Puryear J.D., Newton R.J., Egilla J.N., Grossi J.A.S. (2001) Mycorrhizal fungi enhance accumulation and tolerance of chromium in sunflower (*Helianthus annuus*), J. Plant Physiol. 158, 777–786.
- Declerck S., Risede J.M., Rufyikiri G., Delvaux B. (2002) Effects of arbuscular mycorrhizal fungi on severity of root rot of bananas caused by *Cylindrocladium spathiphylli*, Plant Pathol. 51, 109–115.
- Dickson S., Smith F.A., Smith S.E. (2007) Structural differences in arbuscular mycorrhizal symbioses: more than 100 years after Gallaud, where next? Mycorrhiza 17, 375–393.
- Dodd J.C. (2000) The role of arbuscular mycorrhizal fungi in agro- and natural ecosystems, Outlook Agr. 29, 63–70.
- Eckardt N.A. (2005) Insights into plant cellular mechanisms: of phosphate transporters and arbuscular mycorrhizal infection, Plant Cell 17, 3213–3216.
- Elsen A., Declerck S., De Waele D. (2001) Effects of *Glomus* intraradices on the reproduction of the burrowing nematode (*Radopholus similis*) in deixenic culture, Mycorrhiza 11, 49–51.
- Engvild K. (1987) Nodulation and nitrogen fixation mutants of pea, *Pisum sativum L.*, Theor. Appl. Genet. 74, 711–713.

- Feng G., Zhang F.S., Li X.L., Tian C.Y., Tang C., Rengel Z. (2002) Improved tolerance of maize plants to salt stress by arbuscular mycorrhiza is related to higher accumulation of soluble sugars in roots, Mycorrhiza 12, 185–190.
- Finlay R.D. (2008) Ecological aspects of mycorrhizal symbiosis: with special emphasis on the functional diversity of interactions involving the extraradical mycelium, J. Exp. Bot. 59, 1115–1126.
- Finlay R.D., Rosling A. (2006) Integrated nutrient cycles in forest ecosystems, the role of ectomycorrhizal fungi, in: Gadd G.M. (Ed.), Fungi in biogeochemical cycles, Cambridge, UK, Cambridge University Press, pp. 28–50.
- Finlay R.D., Lindahl B.D., Taylor A.F.S. (2008) Responses of mycorrhizal fungi to stress, in: Avery S., Stratford M., van West P. (Eds.), Stress in yeasts and filamentous fungi, Amsterdam, Elsevier, pp. 201–220.
- Frank B. (1885) Über die auf Wurzelsymbiosen beruhende Ernährung gewisser Bäume durch unterirdische Pilze, Berichte der Deutschen Botanischen Gesellschaft 3, 128–145.
- Franken P., Donges K., Grunwald U., Kost G., Rexer K.-H., Tamasloukt M., Waschke A., Zeuske D. (2007) Gene expression analysis of arbuscule development and functioning, Phytochemistry 68, 68–74.
- Galli U., Schuepp H., Brunold C. (1995) Thiols of Cu-treated maize plants inoculated with the arbuscular mycorrhizal fungus *Glomus intraradices*, Physiol. Plantarum 94, 247–253.
- Gange A.C., Brown V.K., Aplin D.M. (2003) Multitrophic links between arbuscular mycorrhizal fungi and insect parasitoids, Ecol. Lett. 6, 1051–1055.
- Garcia-Garrido J.M., Ocampo J.A. (2002) Regulation of the plant defence response in arbuscular mycorrhizal symbiosis, J. Exp. Bot. 53, 1377–1386.
- Garcia V.I., Mendoza R.E. (2007) Arbuscular mycorrhizal fungi and plant symbiosis in a saline-sodic soil, Mycorrhiza 17, 167–174.
- Garg N., Geetanjali, Kaur A. (2006) Arbuscular mycorrhizal: Nutritional aspects, Arch. Agron. Soil Sci. 52, 593–606.
- Gaur A., Adholeya A. (2004) Prospects of arbuscular mycorrhizal fungi in phytoremediation of heavy metal contaminated soils, Curr. Sci. 86, 528–534.
- Ghazi N., Al-Karaki G.N. (2006) Nursery inoculation of tomato with arbuscular mycorrhizal fungi and subsequent performance under irrigation with saline water, Sci. Hortic. 109, 1–7.
- Genre A., Chabaud M., Faccio A., Barker D.G., Bonfante P. (2008) Prepenetration apparatus assembly precedes and predicts the colonization patterns of arbuscular mycorrhizal fungus within the root cortex of both *Medicago truncatula* and *Daucus carota*, Plant Cell 20, 1407–1420.
- Genre A., Chabaud M., Timmers T., Bonfante P., Barker D.G. (2005) Arbuscular mycorrhizal fungi elicit a novel intracellular apparatus in *Medicago truncatula* root epidermal cells before infection, Plant Cell 17, 3489–3499.
- Gianinazzi S., Schüepp H. (1994) Impact of arbuscular mycorrhizas on sustainable agriculture and natural ecosystems, Birkhäuser Verlag, Basel, p. 226.
- Gianinazzi S., Schüepp H., Barea J.M., Haselwandter K. (2002) Mycorrhizal technology in agriculture – From genes to bioproducts, Birkhäuser, Basel.

- Gianinazzi-Pearson V., Gianinazzi S. (1976) Enzymatic studies on the metabolism of vesicular-arbuscular mycorrhiza. I. Effect of mycorrhiza formation and phosphorus nutrition on soluble phosphatase activities in onion roots, Physiol. Veg. 14, 833–841.
- Gianinazzi-Pearson V., Arnould C., Outfattole M., Arango M., Gianinazzi S. (2000) Differential activation of H+-ATPase genes by an arbuscular mycorrhizal fungus in root cells of transgenic tobacco, Planta 211, 609–613.
- Giovannetti M., Avio L., Fortuna P., Pelligrino E., Sbrana C., Strani P. (2006) At the root of the wood wide web: self recognition and non-self incompatibility in mycorrhizal networks, Plant Signaling and Behavior 1, 1–5.
- Giovannetti M., Avio L., Sbrana C., Citernesi A. (1993a) Factors affecting appressoria development in the vesiculararbuscular mycorrhizal fungus *Glomus mosseae* (Nicol. & Gerd.) Gerd. and Trappe, New Phytol. 123, 114–122.
- Giovannetti M., Sbrana C., Avio L., Citernesi A., Logi C. (1993b) Differential hyphal morphogenesis in arbuscular mycorrhizal fungi during pre-infection stages, New Phytol. 125, 587–593.
- Giovannetti M., Sbrana C., Citernesi A.S., Avio L. (1996) Analysis of factors involved in fungal recognition responses to host derived signals by arbuscular mycorrhizal fungi, New Phytol. 133, 65–71.
- Giri B., Mukerji K.J. (2004) Mycorrhizal inoculant alleviates salt stress in *Sesbania aegyptiaca* and *Sesbania grandiflora* under field conditions: Evidence for reduced and improved magnesium uptake, Mycorrhiza 14, 307–312.
- Giri B., Kapoor R., Mukerji K.G. (2003) Influence of arbuscular mycorrhizal fungi and salinity on growth, biomass and mineral nutrition of *Acacia auriculiformis*, Biol. Fert. Soils 38, 176–180.
- Giri B., Kapoor R., Mukerji K.G. (2007) Improved tolerance of Acacia nilotica to salt stress by arbuscular mycorrhiza, Glomus fasciculatum may be partly related to elevated K/Na ratios in root and shoot tissues, Microbial Ecol. 54, 753–760.
- Glassop D., Smith S.E., Smith F.W. (2005) Cereal phosphate transporters associated with the mycorrhizal pathway of phosphate uptake into roots, Planta 222, 688–698.
- Glazebrook J. (2005) Contrasting mechanisms of defense against biotrophic and necrotrophic pathogens, Annu. Rev. Phytopathol. 43, 205–227.
- Gonzalez-Chavez M.C., Carrillo-Gonzalez R., Wright S.F., Nichols K. (2004) The role of glomalin, a protein produced by arbuscular mycorrhizal fungi, in sequestering potentially toxic elements, Environ. Pollut. 130, 317–323.
- Govindarajulu M., Pfeffer P.E., Jin H., Abubaker J., Douds D.D., Allen J.W., Bucking H., Lammers P.J., Shachar-Hill Y. (2005) Nitrogen transfer in the arbuscular mycorrhizal symbiosis, Nature 435, 819–823.
- Graham J.H., Syversen J.P. (1984) Influence of vesicular arbuscular mycorrhiza on the hydraulic conductivity of roots of two citrus rootstocks, New Phytol. 97, 277–284.
- Guerrieri E., Lingua G., Digillo M.C., Massa N., Berta G. (2004) Do interactions between plant roots and the rhizosphere affect parasitoid behaviour? Ecol. Entomol. 29, 753–756.
- Guimil S., Chang H.S., Zhu T., Sesma A., Osbourn A., Roux C., Ionnidis V., Oakeley E., Docquier M., Descombes P., Briggs S., Paszkowski U. (2005) Comparative transcriptomics of

rice reveals an ancient pattern of response to microbial colonization, Proc. Natl. Acad. Sci. USA 102, 8066–8070.

- Harrison M. (2005) Signaling in the arbuscular mycorrhizal symbiosis, Annu. Rev. Microbiol. 59, 19–42.
- Harrison M.J., van Buuren M.L. (1995) A phosphate transporter from the mycorrhizal fungus *Glomus versiforme*, Nature 378, 626–629.
- Harrison M.J., Dewbre G.R., Liu J.Y. (2002) A phosphate transporter from *Medicago truncatula* involved in the acquisiton of phosphate released by arbuscular mycorrhizal fungi, Plant Cell 14, 2413–2429.
- Haury G., Schikarski W. (1997) Radioactive inputs into the environment; comparison of natural and man-made inventories, in: Stumm W. (Ed.), Global chemical cycles and their alternations by man, Dahlem Konferenzen, Berlin, Germany, pp. 165–188.
- Hawkins H.J., Johansen A., George E. (2000) Uptake and transport of organic and inorganic nitrogen by arbuscular mycorrhizal fungi, Plant Soil 226, 275–285.
- Hayes W.J., Chaudhry T.M., Buckney R.T., Khan, A.G. (2003) Phytoaccumulation of trace metals at the Sunny Corner mine, New South Wales, with suggestions for a possible remediation strategy, Aust. J. Toxicol. 9, 69–82.
- He X., Nara K. (2007) Element biofortification: can mycorrhizas potentially offer a more effective and sustainable pathway to curb human malnutrition? Trends Plant Sci. 12, 331–333.
- He X.H., Critchley C., Bledsoe C. (2003) Nitrogen transfer within and between plants through common mycorrhizal networks (CMNs), Crit. Rev. Plant Sci. 22, 531–567.
- He Z.L., Yang X.E., Stoffella P.J. (2005) Trace elements in agroecosystems and impacts on the environment, J. Trace Elem. Med. Bio. 19, 125–140.
- Heijden M.V.D., Klironomos J., Ursic M., Moutoglis P., Streitwolf-Engel R., Boller T., Wiemken A., Sanders I. (1998). Mycorrhizal fungal diversity determines plant biodiversity, ecosystem variability and productivity, Nature 396, 69–72.
- Helgason T., Fitter A. (2009) Natural selection and the evolutionary ecology of the arbuscular mycorrhizal fungi (Phylum Glomeromycota), J Exp. Bot. 60, 2465–2480.
- Hijri M., Sanders I.R. (2004) The arbuscular mycorrhizal fungus *Glomus intraradices* is haploid and has a small genome size in the lower limit of eukaryotes, Fungal Genet. Biol. 41, 253–261.
- Hodge A., Campbell C.D., Fitter A.H. (2001) An arbuscular mycorrhizal fungus accelerates decomposition and acquires nitrogen directly from organic material, Nature 413, 297–299.
- Hohnjec N., Vieweg M.F., Puhler A., Becker A., Kuster H. (2005) Overlaps in the transcriptional profiles of *Medicago truncatula* roots inoculated with two different *Glomus* fungi provide insights into the genetic program activated during arbuscular mycorrhiza, Plant Physiol. 137, 1283–1301.
- Hooker J.E., Jaizme-Vega M., Atkinson D. (1994) Biocontrol of plant pathogens using arbuscular mycorrhizal fungi, in: Gianinazzi S., Schüepp H. (Eds.), Impact of Arbuscular Mycorrhizas on Sustainable Agriculture and Natural Ecosystems, Basel, Switzerland, Birkhäuser-Verlag, pp. 191–200.
- Hosny M., Gianinazzi-Pearson V., Dulieu H. (1998) Nuclear DNA contents of eleven fungal species in Glomales, Genome 41, 422–428.

- Hosny M., Pais de Barros J.P., Gianinazzi-Pearson V., Dulieu H. (1997) Base composition of DNA from glomalean fungi: high amounts of methylated cytosine, Fungal Genet. Biol. 22, 103–111.
- Hughes J.K., Hodge A., Fitter A.H., Atkin O.W. (2008) mycorrhizal respiration: implications for global scaling relationships, Trends Plant Sci. 13, 583–588.
- Jackson L.E., Burger M., Cavagnaro T.R. (2008) Nitrogen transformations and ecosystem services, Annu. Rev. Plant Biol. 59, 341–363.
- Jaizme-Vega M.C., Azcón R. (1995) Response of some tropical and subtropical cultures to endomycorrhizal fungi, Mycorrhiza 5, 213–217.
- Jaizme-Vega M.C., Sosa Hernández B., Hernández J.M. (1998) Interaction of arbuscular mycorrhizal fungi and the soil pathogen *Fusarium oxysporum* f.sp. *cubense* on the first stages of micropropagated Grande Naine banana, Acta Hortic. 490, 285–295.
- Jaizme-Vega M.C., Tenoury P., Pinochet J., Jaumot M. (1997) Interactions between the root-knot nematode *Meloidogyne* incognita and *Glomus mosseae* in banana, Plant Soil 196, 27–35.
- Jakobsen I., Chen B.D., Munkvold L., Lundsgaard T., Zhu Y.-G. (2005) Contrasting phosphate acquisition of mycorrhizal fungi with that of root hairs using the root hairless barley mutant, Plant Cell Environ. 28, 928–938.
- Jamal A., Ayub N., Usman M., Khan A.G. (2002) Arbuscular mycorrhizal fungi enhance zinc and nickel uptake from contaminated soil by soybean and lentil, Int. J. Phytoremed. 4, 205–221.
- Javot H., Lauvergeat V., Santoni V., Martin-Laurent F., Güclü J., Vinh J., Heyes J., Franck K.I., Schäffner A.R., Bouchez D., Maurel C. (2003) Role of a single aquaporin isoform in root water uptake, Plant Cell 15, 509–522.
- Javot H., Penmesta R.V., Terzaghi N., Cook D.R., Harrison M.J. (2007) A *Medicago truncatula* phosphate transporter indispensable for the arbuscular mycorrhizal symbiosis, PNAS 104, 1720.
- Jeffries P., Gianinazzi S., Perotto S., Turnau K., Barea J.M. (2003) The contribution of arbuscular mycorrhizal fungi in sustainable maintenance of plant health and soil fertility, Biol. Fert. Soils 37, 1–16.
- Jin H., Pfeffer P.E., Douds D.D., Piotrowski E., Lammers P.J., Shachar-Hill Y. (2005) The uptake, metabolism, transport and transfer of nitrogen in an arbuscular mycorrhizal symbiosis, New Phytol. 168, 687–696.
- Johansen A., Finlay R.D., Olsson P.A. (1996) Nitrogen metabolism of external hyphae of the arbuscular mycorrhizal fungus *Glomus intraradices*, New Phytol. 133, 705–712.
- Johansen A., Jakobsen I., Jensen E.S. (1992) Hyphal transport of <sup>15</sup>N labelled nitrogen by a vesicular-arbuscular mycorrhizal fungus and its effect on depletion of inorganic soil N, New Phytol. 122, 281–288.
- Johansen A., Jakobsen I., Jensen E.S. (1993) Hyphal transport by vesicular-arbuscular mycorrhizal fungus on N applied to the soil as ammonium or nitrate, Biol. Fert. Soils 16, 66–70.
- Johansson J.F., Paul L.R., Finlay R.D. (2004) Microbial interactions in the mycorrhizosphere and their significance for sustainable agriculture, FEMS Microbiol. Ecol. 48, 1–12.

- Johnson D., Leake J.R., Ostle N., Ineson P., Read D.J. (2002) In situ CO pulse-labelling of upland grassland demonstrates that a rapid pathway of carbon flux from arbuscular mycorrhizal mycelia to the soil, New Phytol. 153, 327–334.
- Joner E.J., Briones R., Levyal C. (2000) Metal-binding capacity of arbuscular mycorrhizal mycelium, Plant Soil 226, 227–234.
- Journet E.P., El-Gachtouli N., Vernoud V., de Billy F., Pichon M., Dedieu A., Arnould C., Morandi D., Barker D.G., Gianinazzi-Pearson V. (2001) *Medicago truncatula ENOD11*: A novel RPRP-encoding early nodulin gene expressed during mycorrhization in arbuscule-containing cells, Mol. Plant Microbe Int. 14, 737–748.
- Juniper S., Abbott L.K. (2006) Soil salinity delays germination and limits growth of hyphae from propagules of arbuscular mycorrhizal fungi, Mycorrhiza 16, 371–379.
- Kaldorf M., Kuhn A.J., Schroder W.R., Hilderbrandt U., Bothe H. (1999) Selective element deposits in maize colonized by a heavy metal tolerance conferring arbuscular mycorrhizal fungus, J. Plant Physiol. 154, 718–728.
- Karandashov V., Bucher M. (2005) Symbiotic phosphate transport in arbuscular mycorrhizas, Trends Plant Sci. 10, 22–29.
- Khalvati M.A., Hu Y., Mozafar A., Schmidhalter U. (2005) Quantification of water uptake by arbuscular mycorrhizal hyphae and its significance for leaf growth, water relations, and gas exchange of barley subjected to drought stress, Plant Biology 7, 706–712.
- Khan A.G. (2001) Relationships between chromium biomagnification ratio, accumulation factor, and mycorrhizae in plants growing on tannery effluent-polluted soil, Environ. Int. 26, 417–423.
- Khan A.G. (2006) Mycorrhizoremediation an enhanced form of phytoremediation, J. Zhejiang Univ. Sci. B 7, 503–514.
- Kistner C., Winzer T., Pitzschke A., Mulder L., Sato S., Kaneko T., Tabata S., Sandal N., Stougaard J., Webb K.J., Szczyglowski K., Parniske M. (2005) Seven *Lotus japonicus* genes required for transcriptional reprogramming of the root during fungal and bacterial symbiosis, Plant Cell 17, 2217–2229.
- Kosuta S., Chabaud M., Lougnon G., Gough C., Denarie J., Barker D.G., Becard G. (2003) A diffusible factor from arbuscular mycorrhizal fungi induces symbiosis-specific *MtENOD11* expression in roots of *Medicago truncatula*, Plant Physiol. 131, 952–962.
- Krajinski F., Hause B., Gianinazzi-Pearson V., Franken P. (2002) *Mtha1*, a plasma membrane H+-ATPase gene from *Medicago truncatula*, shows arbuscule-specific induced expression in mycorrhizal tissue, Plant Biology 4, 754–761.
- Kruse E., Uehlein N., Kaldenhoff R. (2006) The aquaporins, Genome Biol. 7, 206.
- Lanfranco L., Bolchi A., Ros E.C., Ottonello S., Bonfante P. (2002) Differential expression of a metallothionein gene during the presymbiotic versus the symbiotic phase of an arbuscular mycorrhizal fungus, Plant Physiol. 130, 58–67.
- Leake J., Johnson D., Donnely D., Muckle G., Boddy L., Read D. (2004) Networks of power and influence: the role of mycorrhizal mycelium in controlling plant communities and agroecosystem functioning, Can. J. Bot. 82, 1016–1045.
- Leigh J., Hodge A., Fitter A.H. (2009) Arbuscular mycorrhizal fungi can transfer substantial amounts of nitrogen to their host plant from organic material, New Phytol. 181, 199–207.

- Levyal C., Joner E.J. (2000) Bioavailability of metals in the mycorrhizosphere, in: Gorban G.R., Wenzel W.W., Lombi E. (Eds.), Trace elements in the rhizosphere, CRC, Boca Raton, USA, pp. 165–185.
- Leyval C., Turnau K., Haselwandter K. (1997) Effect of heavy metal pollution on mycorrhizal colonization and function – physiological, ecological and applied aspects, Mycorrhiza 7, 139–153.
- Levyal C., Joner E.J., del Val C., Haselwandter K. (2002) Potential of arbuscular mycorrhizal fungi for bioremediation, in: Gianinazzi S., Schuepp H., Barea J.M., Haselwandter K. (Eds.), Mycorrhizal technology in agriculture: From genes to bioproducts, Birkhauser Verlag, Basel, pp. 175–186.
- Liao J.P., Lin X.G., Cao Z.H., Shi Y.Q., Wong M.H. (2003) Interactions between arbuscular mycorrhizae and heavy metals under sand culture experiment, Chemosphere 50, 847–853.
- Linderman R.G. (1994) Role of VAM fungi in biocontrol, in: Health Pfleger F.L., Linderman R.G. (Eds.), Mycorrhizae and plants, American Phytopathological Society, St. Paul, pp. 1–27.
- Lingua G., D'Agostino G., Massa N., Antosiano M., Berta G. (2002) Mycorrhiza-induced differential response to a yellows disease in tomato, Mycorrhiza 12, 191–198.
- Liu J., Blaylock L.A., Endre G., Cho J., Town C.D., Van den Bosch K.A., Harrison M.J. (2003) Transcript profiling coupled with spatial expression analyses reveals genes involved in distinct developmental stages of an arbuscular mycorrhizal symbiosis, Plant Cell 15, 2106–2123.
- Logi C., Sbrana C., Giovannetti M. (1998) Cellular events involved in survival of individual arbuscular mycorrhizal symbionts growing in the absence of the host, Appl. Environ. Microb. 64, 3473–3479.
- Luu D.-T., Maurel C. (2005) Aquaporins in a challenging environment: molecular gears for adjusting plant water status, Plant Cell Environ. 28, 85–96.
- Maeda D., Ashida K., Iguchi K., Chechetka S.A., Hijikata A., Okusako Y., Deguchi Y., Izui K., Hata S. (2006) Knockdown of an arbuscular mycorrhiza-inducible phosphate transporter gene of *Lotus japonicus* suppresses mutualistic symbiosis, Plant Cell Physiol. 47, 807–817.
- Maldonado-Mendoza I.E., Dewbre G.R., Harrison M.J. (2001) A phosphate transporter gene from the extra-radical mycelium of an arbuscular mycorrhizal fungus *Glomus intraradices* is regulated in response to phosphate in the environment, Mol. Plant Microbe In. 14, 1140–1148.
- Manchanda G., Garg N. (2007) Endomycorrhizal and rhizobial symbiosis: How much do they share? J. Plant Interact. 2, 79–88.
- Marschner H. (1995) Mineral nutrition of higher plants. London, UK, Academic Press.
- Marschner H., Dell B. (1994) Nutrient uptake in mycorrhizal symbiosis, Plant Soil 159, 89–102.
- Marulanda A., Azcón R., Ruiz-Lozano J.M. (2003) Contribution of six arbuscular mycorrhizal fungal isolates to water uptake by *Lactuca sativa* plants under drought stress, Physiol. Plant 119, 526–533.
- Matusova R., Rani K., Verstappen F.W., Franssen M.C., Beale M.H., Bouwmeester H.J. (2005) The strigolactone germination stimulants of the plant-parasitic *Striga* and *Orobanche* spp. are derived from the carotenoid pathway, Plant Physiol. 139, 920–934.

- McGonigle T.P., Miller M.H. (1996) Development of fungi below ground in association with plants growing in disturbed and undisturbed soils, Soil Biol. Biochem. 28, 263–269.
- Meharg A.A. (2003) The mechanistic basis of interactions between mycorrhizal associations and toxic metal cations, Mycol. Res. 107, 1253–1265.
- Mosse B. (1988) Some studies relating to "independent" growth of vesicular–arbuscular endophytes, Can. J. Bot. 66, 2533–2540.
- Muchovej R.M. (2004) Importance of mycorrhizae for agricultural crops. SS-AGR-170, Agronomy Department, Florida Cooperative Extension Service, Institute of Food and Agricultural Sciences, University of Florida.
- Nagy R., Karandashov V., Chague V., Kalinkevich K., Tamasloukht M., Xu G., Jakobsen I., Levy A.A., Amrhein N., Bucher M. (2005) The characterization of novel mycorrhizaspecific phosphate transporters from *Lycopersicon esculentum* and *Solanum tuberosum* uncovers functional redundancy in symbiotic phosphate transport in solanaceous species, Plant J. 42, 236–250.
- Nayyar A.A., Hamel C., Hanson K., Germida J. (2008) The arbuscular mycorrhizal symbiosis lins N mineralization to plant demand, Mycorrhiza 19, 239–246.
- Nelsen C.E. (1987) The water relations of vesicular-arbuscular mycorrhizal systems, in: Safir G.R. (Ed.), Ecophysiology of VA mycorrhizal plants, CRC Press, Boca Raton, Fla, pp. 71–79.
- Nelsen C.E., Safir G.R. (1982) Increased drought tolerance of mycorrhizal onion plants caused by improved phosphorus nutrition, Planta 154, 407–413.
- Ohtomo R., Saito M. (2005) Polyphosphate dynamics in mycorrhizal roots during colonization of an arbuscular mycorrhizal fungus, New Phytol. 167, 571–578.
- Olah B., Briere C., Becard G., Denarie J., Gough C. (2005) Nod factors and a diffusible factor from arbuscular mycorrhizal fungi stimulate lateral root formation in *Medicago truncatula* via the DMI1/DMI2 signalling pathway, Plant J. 44, 195–207.
- Osonubi O., Mulongoy K., Awotoye O.O., Atayese M.O., Okali D.U. (1991) Effects of ectomycorrhizal and vesiculararbuscular mycorrhizalfungi on drought tolerance of four leguminous woody seedlings, Plant Soil 136, 131–143.
- Pacovsky R.S., De Silva P., Carvalho M.T.V., Tsai S.M. (1990) Increased nutrient assimilation and enzyme activities in field beans inoculated with *Glomus etunicatum*, Proceedings of the Eight North American Conference on Mycorrhiza, p. 203.
- Pacovsky R.S., De Silva P., Carvalho M.T.V., Tsai S.M. (1991) Growth and nutrient allocation in *Phaseolus vulgaris* L. colonized with endomycorrhizae or *Rhizobium*, Plant Soil 132, 127–137.
- Parniske M. (2004) Molecular genetics of the arbuscular mycorrhizal symbiosis, Curr. Opin. Plant Biol. 7, 414–421.
- Parniske M. (2008) Arbuscular mycorrhiza: the mother of plant root endosymbioses, Nature Rev. Microbiol. 6, 763–775.
- Paszkowski U. (2006) A journey through signaling in arbuscular mycorrhizal symbioses. New Phytol. 172, 35–46.
- Paszkowski U., Kroken S., Roux C., Briggs S.P. (2002) Rice phosphate transporters include an evolutionarily divergent gene specifically activated in arbuscular mycorrhizal symbiosis, Proc. Natl. Acad. Sci. USA 99, 13324–13329.

- Pawlowska T.E., Charvat I. (2002) Influence of edaphic and environmental factors on arbuscular mycorrhizae, in: Sharma A.K., Johri B.N. (Eds.), Arbuscular mycorrhizae: interactions in plants, rhizosphere and soils, Science Publishers, Inc., Enfield, N.H., pp. 105–134.
- Pierce F.J., Lal R. (1991) Soil management in the 21st century, in: Lal R., Pierce F.J. (Eds.), Soil Management for Sustainability, Soil and Water Conservation Society of America, Ankeny, pp. 175–179.
- Porcel R., Aroca R., Azcón R., Ruiz-Lozano J.M. (2006) PIP aquaporin gene expression in arbuscular mycorrhizal *Glycine max* and *Lactuca sativa* plants in relation to drought stress tolerance, Plant Mol. Biol. 60, 389–404.
- Pozo M.J., Slezack-Deschaumes S., Dumas-Gaudot E., Gianinazzi S., Azcón-Aguilar C. (2002) Plant defense responses induced by arbuscular mycorrhizal fungi, in: Gianinazzi S., Schuepp H., Barea J.M. (Eds.), Mycorrhizal Technology in Agriculture, Birkhauser, Basel, pp. 103–112.
- Pumplin N., Harrison M.J. (2009) Live-cell imaging reveals periarbuscular membrane domains and organelle location in *Medicago truncatula* roots during arbuscular mycorrhizal symbiosis, Plant Physiol. Preview.
- Rabie G.H. (2005) Contribution of arbuscular mycorrhizal fungus to red kidney and wheat plants tolerance grown in heavy metal-polluted soil, Afr. J. Biotechnol. 4, 332–345.
- Rabie G.H., Almadini A.M. (2005) Role of bioinoculants in development of salt-tolerance of *Vicia faba* plants under salinity stress, Afr. J. Biotechnol. 4, 210–223.
- Reagnold J.P., Papendick R.I., Parr J.F. (1990) Sustainable agriculture, Science America 262, 112–120.
- Redecker D., Kodner R., Graham L.E. (2000) Glomalean fungi from the Ordovician, Science 289, 1920–1921.
- Requena N., Breuninger M. (2004) The old arbuscular mycorrhizal symbiosis in the light of the molecular era, in: Esser K., Luttge U., Beyschlag W., Murata J. (Eds.), Progress in Botany, Springer, Berlin, Heidelberg, pp. 323–356.
- Requena N., Jeffries P., Barea J.M. (1996) Assessment of natural mycorrhizal potential in a desertified semiarid ecosystem, Appl. Environ. Microb. 62, 842–847.
- Requena N., Mann P., Hampp R., Franken P. (2002) Early developmentally regulated genes in the arbuscular mycorrhizal fungus *Glomus mosseae*: identification of GmGin1 a novel gene with homology to the C-terminus of metazoan hedgehog proteins, Plant Soil 244, 129–139.
- Requena N., Mann P., Franken C. (2000) A homologue of the cell-cycle check-point TOR2 from yeast exist in the arbuscular mycorrhizal fungus *Glomus mosseae*, Protoplasma 211, 89–98.
- Requena N., Serrano E., Ocón A., Breuninger M. (2007) Plant signals and fungal perception during arbuscular mycorrhizal establishment, Phytochemistry 68, 33–40.
- Rosendahl C.N., Rosendahl S. (1991) Influence of vesiculararbuscular mycorrhizal fungi (*Glomus* spp.) on the response of cucumber (*Cucumis sativus* L.) to salt stress, Environ. Exp. Bot. 31, 313–318.
- Roose T., Fowler A.C. (2004) A mathematical model for water and nutrient uptake by plant root systems, J. Theor. Biol. 228, 173–184.
- Rufyikiri G., Declerck S., Delvaux B., Dufey J.E. (2000) Arbuscular mycorrhizal fungi might alleviate aluminium toxicity in banana plants, New Phytol. 148, 343–352.

- Ruiz-Lozano J.M. (2003) Arbuscular mycorrhizal symbiosis and alleviation of osmotic stress. New perspectives for molecular studies, Mycorrhiza 13, 309–317.
- Ruiz-Lozano J.M., Azcón R., Gomez M. (1995) Effects of arbuscular-mycorrhizal *Glomus* species on drought tolerance: physiological and nutritional responses, Appl. Environ. Microb. 61, 456–460.
- Saleh M., Saleh Al-Garni (2006) Increased heavy metal tolerance of cowpea plants by dual inoculation of an arbuscular mycorrhizal fungi and nitrogen-fixer *Rhizobium* bacterium, Afr. J. Biotechnol. 5, 132–144.
- Sannazzaro A.I., Ruiz O.A., Alberto E.O., Menendez A.B. (2006) Alleviation of salt stress in *Lotus glaber* by *Glomus intraradices*, Mycorrhiza 285, 279–287.
- Schultz C.J., Harrison M.J. (2008) Novel plant and fungal AGP-like proteins in the *Medicago truncatula-Glomus intraradices* arbuscular mycorrhizal symbiosis, Mycorrhiza 18, 403–412.
- Selvaraj T. (1998) Studies on mycorrhizal and rhizobial symbioses on tolerance of tannery effluent treated *Prosopis juliflora*, Ph.D. Thesis, University of Madras, Chennai, India.
- Selvaraj T., Chelleppan P. (2006) Arbuscular mycorrhizae: a diverse personality, Central Eur. J. Agr. 7, 349–358.
- Selvaraj T., Chellappan P., Jeong Y.J., Kim H. (2004) Occurrence of vesicular-arbuscular mycorrhizal (VAM) fungi and their effect on plant growth in endangered vegetations, J. Microbiol. Biotechn. 14, 885–890.
- Sheng M., Tang M., Chen H., Yang B., Zhang F., Huang Y. (2008) Influence of arbuscular mycorrhizae on photosynthesis and water status of maize plants under salt stress, Mycorrhiza 18, 287–296.
- Smit P., Raedts J., Portyanko V., Debelle F., Gough C., Bisseling T., Geurts R. (2005) NSPI of the GRAS protein family is essential for rhizobial Nod factor-induced transcription, Science 308, 1789–1791.
- Smith S.E., Read D.J. (2008) Mycorrhizal Symbiosis, Academic Press, Inc., San Diego, CA.
- Smith S.E., Smith F.A., Jakobsen I. (2003) Mycorrizal fungi can dominate phosphate supply to plants irrespective of growth responses, Plant Physiol. 133, 16–20.
- Smith S.E., Smith F.A., Jakobsen I. (2004) Functional diversity in arbuscular mycorrhizal (AM) symbioses: the contribution of the mycorrhizal P uptake pathway is not correlated with mycorrhizal responses in growth or total P uptake, New Phytol. 162, 511–524.
- Smith S.E., Smith F.A., Nicholas D.J.D. (1985) Activity of glutamine synthetase and glutamate dehydrogenase in *Trifolium* subterraneum L. & Allium cepa L. effects of mycorrhizal infection and phosphate nutrition, New Phytol. 99, 211–227.
- Solaiman M.Z., Ezawa T., Kojima T., Saito M. (1999) Polyphosphates in intraradical and extraradical hyphae of an arbuscular mycorrhizal fungus, *Gigaspora margarita*, Appl. Environ. Microb. 65, 5604–5606.
- Spanu P., Bonfante-Fasolo P. (1998) Cell wall bound peroxidase activity in roots of mycorrhizal *Allium porrum*, New Phytol. 109, 119–124.
- St Arnaud M., Hamel C., Caron M., Fortin J. (1994) Inhibition of *Pythium ultimum* in roots and growth substrate of mycorrhizal *Tagetes patula* colonised with *Glomus intraradices*, Can. J. Plant Pathol. 16, 187–94.

- Stewart B.A., Lal R., El-Swaify S.A. (1991) Sustaining the resource base on an expanding world agriculture, in: Lal R., Pierce F.J. Soil Management for Sustainability, Soil and Water Conservation Society of America, Ankeny, pp. 125–144.
- Sturz A.V., Carter M.R., Johnston H.W. (1997) A review of plant disease, pathogen interactions and microbial antagonism under conservation tillage in temperate humid agriculture, Soil Till. Res. 41, 169–189.
- Tahiri-Alaoui A., Antoniw J.F. (1996) Cloning of genes associated with the colonisation of tomato roots by the arbuscular mycorrhizal fungus *Glomus mosseae*, Agronomie 16, 699–707.
- Tisserant B., Gianninazzi-Pearson V., Gianninazzi S., Gollotte A. (1993) Plant histochemical staining of fungal alkaline phosphatase activity for analysis of efficient arbuscular mycorrhizal infections, Mycol. Res. 97, 245–250.
- Tonin C., Vandenkoornhuyse P., Joner E.J., Straczek J., Levyal C. (2001) Assessment of arbuscular mycorrhizal fungi diversity in the rhizosphere of *Viola calaminaria* and effect of these fungi on heavy metal uptake by clover, Mycorrhiza 10, 161–168.
- Toussaint J.P., St-Arnaud M., Charest C. (2004) Nitrogen transfer and a ssimilation between the arbuscular mycorrhizal fungus *Glomus intraradices* Schenck & Smith and Ri T-DNA roots of *Daucus carota* L. in an in vitro compartmented system, Can. J. Microbiol. 50. 251–260.
- Trotta A., Falaschi P., Cornara L., Minganti V., Fusconi A., Drava G. (2006) Arbuscular mycorrhizae increase the arsenic translocation factor in the As hyperaccumulating fern *Pteris vittata* L., Chemosphere 65, 74–81.
- Trotta A., Varese G.C., Gnavi E., Fusconi A., Sampo S., Berta G. (1996) Interactions between the soil-borne root pathogen *Phytophthora nicotianae* var. *parasitica* and the arbuscular mycorrhizal fungus *Glomus mosseae* in tomato plants, Plant Soil 185, 199–209.
- Turk M.A., Assaf T.A., Hammed K.M., Al-Tawaha A.M. (2008) Significance of mycorrhizae, World J. Agric. Sci. 2, 16–20.
- van Dam N.M., Harvey J.A., Wakers F.L., Bezemer T.M., van der Putten W.H., Vet L.E.M. (2000) Interactions between aboveground and belowground induced responses against phytophages, Basic Appl. Ecol. 4, 63–77.
- van der Putten W.H., Klironomos J.H., Wardle D.A. (2007) Microbial ecology of biological invasions, ISME J. 1, 28–37.
- van Rhijn P., Fang Y., Galili S., Shaul O., Atzmon N., Wininger S., Eshed Y., Lum M., Li Y., To V.V., Fujishige N., Kapulnik Y., Hirsch A.M. (1997) Expression of early nodulin genes in alfalfa mycorrhizae indicates that signal transduction pathways used in forming arbuscular mycorrhizae and rhizobium-induced nodules may be conserved, Proc. Natl. Acad. Sci. USA 94, 5467–5472.
- Vazquez M.M., Barea J.M., Azcón R. (2001) Impact of soil nitrogen concentration on *Glomus* spp.–*Sinorhizobium* interactions as affecting growth, nitrate reductase activity and protein content of *Medicago sativa*, Biol. Fert. Soils 34, 57–63.
- Vieweg M.F., Fruhling M., Quandt H.J., Heim U., Baumlein H., Puhler A., Kuster H., Andreas M.P. (2004) The promoter of the *Vicia faba* L. leghemoglobin gene *VfLb29* is specifically activated in the infected cells of root nodules and in the arbuscule-containing cells of mycorrhizal roots from different legume and nonlegume plants, Mol. Plant Microbe In. 17, 62–69.

- Vivas A., Barea J.M., Azcon R. (2005) Interactive effect of *Brevibacillus brevis* and *Glomus mosseae* both isolated from Cd contaminated soil, on plant growth, physiological mycorrhizal fungal characteristics and soil enzymatic activities in Cd polluted soils, Environ. Pollut. 134, 257–266.
- Wallander H. (2006) Mineral dissolution by ectomycorrhizal fungi, in: Gadd G.M. (Ed.), Fungi in biogeochemical cycles, Cambridge University Press, Cambridge, UK, pp. 28–50.
- Wang B., Qiu Y.L. (2006) Phylogenetic distribution and evolution of mycorrhizas in land plants, Mycorrhiza 16, 299–363.
- Wang F., Lin X., Yin R. (2005) Heavy metal uptake by arbuscular mycorrhizas of *Elsholtzia splendens* and the potential for phytoremediation of contaminated soil, Plant Soil 269, 225–232.
- Weissenhorn I., Levyal C., Belgy G., Berthelin J. (1995) Arbuscular mycorrhizal contribution to heavy metal uptake by maize (*Zea mays L.*) in pot culture with contaminated soil, Mycorrhiza 5, 245–251.
- Wetterauer D.G., Killorn R.J. (1996) Fallow- and flooded-soil syndromes: effects on crop production, J. Prod. Agric. 9, 39–41.
- White P.J., Bradley M.R. (2000) Mechanisms of cesium uptake by plant, New Phytol. 147, 241–256.
- Whitefield L., Richards A.J., Rimmer D.L. (2004) Effects of mycorrhizal colonization on *Thymus polytrichus* from heavymetal-contaminated sites in northern England, Mycorrhiza 14, 47–54.
- Wu F.Y., Ye Z.H., Wu S.C., Wong M.H. (2001) Metal accumulation and arbuscular mycorrhizal status in metallicolous and nonmetallicolous populations of *Pteris vittata* L. and *Sedum alfredi* Hance, Planta 226, 43–56.
- Xavier I.J., Boyetchko S.M. (2002) Arbuscular Mycorrhizal Fungi as Biostimulants and Bioprotectants of Crops, in: Khachatourians G.G., Arora D.K. (Eds.), Appl. Mycol. Biotechnol., Vol. 2: Agriculture and Food Production, Elsevier, Amsterdam, pp. 311–330.
- Yano-Melo A.M., Saggin O.J., Lima-Filho J.M., Melo N.F., Maia L.C. (1999) Effect of arbuscular mycorrhizal fungi on the acclimatization of micropropagated banana plants, Mycorrhiza 9, 119–123.
- Yano-Melo A.M., Saggin O.J., Maia L.C. (2003) Tolerance of mycorrhized banana (*Musa* sp. cv. Pacovan) plantlets to saline stress, Agr. Ecosyst. Environ. 95, 343–348.
- Young N.D., Mudge J., Ellis T.H. (2003) Legume genomes: more than peas in a pod, Curr. Opin. Plant Biol. 6, 199–204.
- Yu X., Cheng J., Wong M.H. (2004) Earthworm-mycorrhiza interaction on Cd uptake and growth of ryegrass, Soil Biol. Biochem. 37, 1–7.
- Zandavalli R.B., Dillenburg L.R., Paulo V.D. (2004) Growth responses of *Araucaria angustifolia* (Araucariaceae) to inoculation with the mycorrhizal fungus *Glomus clarum*, Appl. Soil Ecol. 25, 245–255.
- Zhu H., Reily B.K., Burns N.J., Ane J. (2006) Tracing Nonlgume Orthologs of Legume genes required for nodulation and Arbuscular Mycorrhizal Symbioses, Genetics 172, 2491–2499.
- Zhu J.K. (2001) Plant salt tolerance, Trends Plant Sci. 6, 66–71.
- Zhu Y.G., Smolders E. (2000) Plant uptake of radiocesium: A review of mechanisms, regulation and application, J. Exp. Bot. 51, 1635–1645.

# **Efficient N Management Using Winter Oilseed Rape**

Klaus Sieling and Henning Kage

### Contents

1	Introduction	932		
2	N Fertilization in Fall	934		
3	N Fertilization in Spring	935		
4	N Dynamic After Oilseed Rape	938		
5	Conclusion	939		
References				

K. Sieling (🖂)

Institute of Crop Science and Plant Breeding, Christian-Albrechts-University, Hermann-Rodewald-Str. 9, 24118 Kiel, Germany e-mail: sieling@pflanzenbau.uni-kiel.de Abstract During the last decades the acreage of winter oilseed rape has been increased considerably in Europe. Rapeseed can take up a large amount of nitrogen before winter (>100 kg N/ha) and thus prevent nitrate leaching and pollution. Winter wheat is often grown subsequently, using oilseed rape as a favorable preceding crop. However, under wheat large nitrogen losses via leaching are frequently observed in humid climates during winter, mainly due to high amounts of soil mineral N available in fall and the small N uptake in fall of wheat as a subsequent crop. The low N offtake by the seeds results in a lower N-use efficiency and increases the N surpluses (>90 kg N/ha) compared with winter wheat (c. 40 kg N/ha). In addition, a large soil N pool increases the risk of N<sub>2</sub>O emission, with its impact on climate change. In our review we discuss several options to increase nitrogen-use efficiency in oilseed rape-based cropping systems ranging from optimizing N fertilization practices to options arising from adopted tillage practices and crop rotation. N application in fall normally increases dry matter accumulation and N uptake before winter. However, because of its limited yield effects in most situations, fall N supply also boosts N surpluses. N fertilization in spring exceeding the need of the crop for optimal seed yield increases the risk of N leaching and decreases the farmer's net revenue. Considering the amount of N taken up by the canopy before the first spring application improves the determination of the optimal spring N supply. Measuring canopy N in fall gave the best results. At the cropping system level, time and intensity of soil tillage after the harvest of oilseed rape has concurrent goals of controlling volunteer rape, and achieving a successful establishment of the following crop, but avoiding an increased N mineralization. Changing

E. Lichtfouse et al. (eds.), *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_41,
(c) Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Sieling and Kage, Agron. Sustain. Dev. 30 (2010) 271–279. DOI: 10.1051/agro:2009036

the crop rotation by growing catch crops which prevent N from leaching is very effective in reducing N losses from the system by >40%. However, the economic losses from growing a usually less profitable spring crop probably limit the acceptance by farmers. Despite the problems addressed above, looking at the whole cropping system, oilseed rape is indispensable because of its beneficial effects on yield levels and nitrogen-use efficiency of following cereals, especially wheat, because alternative crops are often not realistic alternatives.

**Keywords** Oilseed rape • *Brassica napus* • N fertilization • N leaching • Seed yield • N balance

### **1** Introduction

The 'Nitrate Directive' (Directive 91/676/EEC) released in 1991 by the European Union aims to reduce water pollution caused or induced by nitrates from agricultural sources and prevent such pollution (European Union, 1991). In addition, the 'Water Framework Directive' (Directive 2000/60/EC) expands the scope of water protection to all waters and sets clear objectives that a 'good ecological and chemical status' must be achieved for all European water bodies by 2015 (European Union, 2000). The Water Frame Directive demands the prevention of deterioration in status and a progressive reduction of pollution. Concerning agricultural production, the impact of nitrogen (N) fertilization on the nitrate concentration of the water bodies becomes a crucial point in this context. Agriculture is the major contributor to nitrate contamination of groundwater (Fraters et al., 1998). For example, the German implementation of the Nitrate Directive ('Düngeverordnung') became effective in 2007 in a revised form and defines for the first time, among other regulations, legal thresholds for the N balance, calculated from N fertilization minus N offtake by the harvest products (Düngeverordnung, 2006). The surplus must not exceed a threshold of 90 kg N/ha on a 3-year average in the years 2006-2008, declining to 60 kg N/ha in 2009–2011 in order to reduce the environmental impact of N fertilization. Purely arable cropping systems with low amounts of organic N inputs are often regarded as low risk in terms of N leaching. The calculations of Henke

et al. (2007), however, revealed that ex post optimized N levels of an oilseed rape – winter wheat – winter barley rotation were near the threshold of 60 kg N/ha, and a current survey of N balances in regions with high percentages of oilseed rape in the crop rotation showed that the N surpluses were often clearly above the acceptable level (Kelm et al., 2007).

N balances on a field scale or in larger areas are often used to estimate the leaching risk (Doluschnitz et al., 1992; Lord et al., 2002; Jansons et al., 2003; Sacco et al., 2003). However, there is much evidence that, in the short term, the link between fertilizer use (except excessive amounts) and nitrate in water is not very direct (Fig. 1) (Sieling and Kage, 2006). The N surplus in itself may not be sufficient to quantitatively determine the amount of N loss via various pathways, because of the complex interactions with other environmental parameters. For example, the large reserve of organic N in soils and vegetation will inevitably contribute nitrate to leaching once the land is tilled (Macdonald et al., 1989; Sylvester-Bradley and Chambers, 1992). N leaching depends on the amount of water percolating through the soil, mainly affected by winter rainfall (Webb et al., 2000), and the N concentration in the leachate. The latter is closely related to the mineral N pool in the soil at the beginning of and/or during the leaching period (Goss et al., 1991). Most of the N leached from arable soils originates from inorganic N present in late summer, fall or early winter when soils start to drain and plant

N leaching (kg N/ha)



**Fig. 1** Relationship between N balance (kg N/ha) and annual N leaching (kg N/ha) during the subsequent leaching period in different preceding crop – crop combinations (1991/92–1999/2000). The EU drinking water threshold (50 ppm nitrate) is given for 300 mm percolation (Sieling and Kage, 2006)

demand is low or non-existent. Generally, N leaching positively correlates with the N supply. Mineral N application has been shown to significantly increase N losses only for N rates exceeding the ex post determined economic optimum (e.g. Engels and Kuhlmann, 1993; Davies and Sylvester-Bradley, 1995; Goulding et al., 2000). Results from the Broadbalk Wheat Experiment in Rothamsted (United Kingdom) indicated that in most years only a little fertilizer-derived N remained in inorganic form at harvest from applications of up to 192 kg N/ha (Glendining et al., 1996). In the short term, therefore, reduction of N fertilization below the optimum had only small effects on fall soil mineral N and, in consequence, on N losses over winter (Macdonald et al., 1989; Lickfett, 1993; Zerulla et al., 1993; Shepherd and Sylvester-Bradley, 1996). More serious problems may arise from the use of organic manures and slurries, which are often applied to the stubbles after harvest on arable land, when no plant uptake occurs. Compared with mineral N fertilizers, crops utilize slurry N poorly, due to overwinter losses by leaching or denitrification (Smith and Chambers, 1992; Sieling, 2005).

On the other hand, however, the simple N balance is often the single parameter that can be estimated at the field, farm and regional level. It can give useful risk indications of specific farming practices, especially in the wider environment and if integrated over longer time periods (Öborn et al., 2003). Even if N fertilization meets plant N requirement in time and rate, long-term application of inorganic N fertilizer and, to a greater extent, of organic manures, may induce the building up of soil organic matter due to increased amounts and higher N concentration of crop residues being returned to the soil at harvest (Glendining et al., 1996). Up to certain levels this may be seen as a positive effect on soil fertility and the overall carbon balance, but it should also be clear that in consequence, N mineralization of soil organic N in the fallow periods, and consequently N leaching, may increase.

Farmers are asked to reduce the impact of nitrogen on the environment. Increased nitrogen supply to crops may increase yield but decreases N-use efficiency (Kuhlmann and Engels, 1989; Sieling and Hanus, 1997). If it has not been volatilized or denitrified, the N amount not utilized by the crop can accumulate in the soil and, in consequence, escalates the risk of leaching with corresponding environmental consequences.

Especially in rotations including winter oilseed rape N leaching is often a great problem (Fig. 1). Indeed, oilseed rape can be a suitable crop to conserve nitrogen throughout the winter, because of its large fall N uptake of 40–60 kg N/ha (Barraclough, 1989; Aufhammer et al., 1994). Due to its early development, oilseed rape often can utilize only a small proportion of the nitrogen mineralized in spring, especially under weather conditions with slow temperature increase in spring (Sieling et al., 1998a). Additional large amounts of easily mineralizable crop residue (petals and leaves) return to the soil after flowering. The harvest index (0.35) and the N-use efficiency of oilseed rape are low compared with cereals (Aufhammer et al., 1994; Shepherd and Sylvester-Bradley, 1996; Malagoli et al., 2005; Rathke et al., 2005; Sieling et al., 1998a, b, 1999; Sieling and Kage, 2006; Henke et al., 2007; Berry, 2009). Moreover, oilseed rape leaves the soil in a favorable structure, leading to an increased N release. All these facts result in large N amounts remaining in the soil (Shepherd and Sylvester-Bradley, 1996; Beaudoin, 2005). In most situations the subsequent crop following rape will be winter wheat (Sieling, 2005). Even sown early, N uptake of wheat before winter normally does not exceed 20 kg N/ha, which is only a small portion of the soil mineral N usually present after oilseed rape. In consequence, most of these N residues cannot be completely taken up by the subsequent wheat crop, and increase the risk of N leaching into the groundwater during the following percolation period (Goss et al., 1993; Sieling and Kage, 2006). Having a crop rotation where each crop is always following the same preceding crop, a differentiation between the preceding crop and the direct crop effects on N leaching is not possible.

Several approaches to reduce N leaching have been discussed, e.g. changes in crop rotation, growing of catch crops, reducing soil tillage in fall and reducing N fertilization (Allison et al., 1998; Kuhlmann and Engels, 1989; Lindén and Walgren, 1993; Rinnofner et al., 2008; Shepherd and Lord, 1996; Smith et al., 1990). However, acceptance by farmers often remains low, mainly due to economic losses.

The aim of this review was to investigate the necessity and the extent of some management measures carried out during (N fertilization in fall, N fertilization in spring) or following (soil tillage after harvest, growing catch crops) the rapeseed crop in order to improve the N-use efficiency and to reduce the environmental impact of oilseed rape production.

### 2 N Fertilization in Fall

Applying fall nitrogen to oilseed rape is common practice in NW Europe. In the last few years farmers have changed the crop rotation, replacing barley by wheat. In consequence, the sowing date is often delayed. In addition, they increasingly pass on plowing during seedbed preparation to reduce energy costs. To compensate for this worsening of establishment conditions, they additionally apply about 30–50 kg N/ha in fall, often directly upon the stubble of the preceding crop, to ensure crop N supply and adequate crop growth before winter.

In general it has to be distinguished if the rapeseed crop is actually able to take up fall-applied N before winter. If low temperature limits crop growth due to delayed sowing dates, applied N increases the soil mineral N (NO<sub>3</sub>–N plus NH<sub>4</sub>–N) pool and, in consequence, the potential of N leaching during the subsequent percolation period (Table 1, sown in the 1st decade of September) (Sieling and Kage, 2007). In contrast, Engström et al. (2009) reported that application of 30 or 60 kg N/ha, at sowing, to winter oilseed rape did not affect the N leaching or seed yield. If fall N, especially in early-sown oilseed rape (Dejoux et al., 2003), leads to a better growth and enhances N accumulation before winter, it is a moot question whether the seed yield increases as well. Ogilvy and Bastiman (1992) reported that, although plots receiving nitrogen in the seedbed or at the two leaf stage appeared more vigorous before winter compared with unfertilized plants, neither the number of plants established, the survival over winter nor the seed yield appeared to be affected by this treatment (see also Table 1, sown in the 1st decade of September). Sieling and Kage (2007) observed after a fall N supply of 40 kg N/ha a yield increase of about 0.2 t/ha which was related to an additional N offtake by the seeds of about 4 kg N/ha. In maximum only 10% of the applied N amount was removed from the system, whereas 36 kg N/ha remained in the soil and charged the N balance (Table 2). Colnenne et al. (2002) and Flénet et al. (2009) observed in their experiments severe N deficiencies in fall, described in terms of the nitrogen nutrition index, together with a reduction in shoot biomass, tap root

Table 1         Effect of fall-applied	Sowing date Fall	Middle of August		Beginning of September		
uptake before winter, soil	N (kg N/ha)	0	40	0	40	
mineral N before winter and seed yield of oilseed rape in	N uptake before winter (kg N/ha)	95	131	65	74	
2005/06 (Henke, 2008)	Soil mineral N content before winter (kg N/ha)	19	23	34	63	
	Seed yield (t/ha)	4.69	4.75	4.80	4.86	

Table 2 Effect of 40 kg N/ha applied in fall on above-ground dry matter before winter, N uptake before winter
seed yield and simple N balance of the fall-applied fertilizer N of oilseed rape (2003/04-2005/06, cv. Talent
minimum tillage following wheat)

	Above-ground dry matter $(g/m^2)$	N uptake (kg N/ha)	Seed yield (t/ha)	N balance (kg N/ha)
Unfertilized control 40 kg N/ha on the wheat	48.2 <sup>b‡</sup>	18.2 <sup>b</sup>	4.59 <sup>b</sup>	_
stubble 40 kg N/ha directly after	68.1 <sup>ab</sup>	24.7 <sup>b</sup>	4.71 <sup>ab</sup>	+36
drilling 40 kg N/ha in the 2-4 leaf	86.6 <sup>a</sup>	36.3 <sup>a</sup>	4.69 <sup>ab</sup>	+37
stage of oilseed rape	87.0 <sup>a</sup>	36.7 <sup>a</sup>	4.81 <sup>a</sup>	+33

‡ Within a column, means followed by the same letter are not significantly different at P < 0.05.

biomass, leaf area index and radiation-use efficiency compared with well supplied treatments. However, despite severe fall N deficiencies, no difference in seed yield was apparent. The authors assumed that all the time allowed enough growth in fall to ensure sufficient regrowth in spring. However, in other experiments fall nitrogen gave a small yield response where the preceding cereal straw was baled or incorporated instead of burning (Chalmers, 1989; Chalmers and Darby, 1992).

Due to its ability to take up substantial quantities of nitrogen before winter (Barraclough, 1989), oilseed rape is regarded as a suitable crop to utilize fall slurry and to prevent slurry nitrogen from leaching during the percolation period. Detailed analysis revealed the higher leaching potential of fall slurry (up to 4.3 times) and, to a smaller extent, of spring slurry (up to 1.7 times) compared with mineral N fertilizers. The larger losses from slurry could be caused by its slower rate of N mineralization and the release of N when the crop cannot use it (Sieling et al., 1998a; Sieling and Kage, 2006). Applying slurry (80 kg N<sub>total</sub>/ha) in spring during crop growth instead of fall allows one to reduce the additional mineral N fertilization by 40 kg N/ha (Fig. 2). However, further details about the effect of slurry applied in fall and/or in spring on seed yield are scarce.

N fertilization of oilseed rape in fall seems to be necessary only if N really limits crop growth, for example on poor soils or in the case of minimum tillage if the seeds germinate within a straw layer. If farmers assume that the rapeseed crop needs N in fall, N fertilizer should be applied to the crop itself and not before sowing (Sieling and Kage, 2007). On the other hand, the



Fig. 2 Effects of slurry application in fall and spring on seed yield of oilseed rape

effect of a late N shortage in fall, e.g. on light soils, on yield potential of an early-sown and well-established rapeseed crop remains unclear.

### **3 N Fertilization in Spring**

An exact estimation of fertilizer N demand has become increasingly important to minimize the environmental impact, but also to achieve high seed yields and maximum economic returns. For this purpose, a better understanding of the course of N uptake and the yield formation is essential. Seed yield of oilseed rape consists of the number of plants/m<sup>2</sup>, the number of pods per plant, the number of seeds per pod and the single seed weight (Grosse et al., 1992; Diepenbrock, 2000; Rathke et al., 2006). The onset of branches and buds starts even in late fall, whereas the final number of pods per plant and seeds per pod are determined within 1-2 weeks after flowering. Fall-sown oilseed rape normally receives nitrogen in spring as a split application at the beginning of growth and at stem elongation. This means that N fertilizers are normally applied before yield formation occurs. In contrast to wheat, it is not possible to directly support yield formation. In addition, some results showed that nitrate uptake increases from stem elongation to the beginning of flowering, whereas nitrate uptake during pod filling was low (Merrien et al., 1988; Jensen et al., 1997; Rossato et al., 2001). Other authors reported that N uptake remained high even after the start of pod filling (Schjoerring et al., 1995; Hocking et al., 1997; Malagoli et al., 2005). Spink (2009) pointed out that late N is required to maintain pod filling. However, most of the N supply in the pods was achieved mainly by N mobilization from vegetative parts (stem, leaves, taproot) (Malagoli et al., 2005).

The spring nitrogen requirement mainly depends on expected yield level and soil type (Chalmers and Darby, 1992). Zhao et al. (1993) suggested a maximum yield response to a N rate around 200 kg N/ha, which was in good agreement with the results of Bilsborrow et al. (1993), who obtained >85% of the maximum recorded yield with an application of 150 kg N/ha. Over the years 1989/90 to 1991/92, Aniol (1993) tested N amounts from 0 to 320 kg N/ha, varying the timing and distribution pattern. Using a multiple regression model he estimated a yield maximum of 5.04 t/ha in plots receiving 308 kg N/ha. Comparing similar amounts of N, distributions with a larger amount at the earlier stages (beginning of growth) increased the seed yield more than high rates at later stages. Split application ensured that high yield performance and optimum yield stability were achieved simultaneously (Boelcke et al., 1991). In contrast, Darby and Hewitt (1990) observed only small effects of timing and distribution of fertilizer N on seed yield. Super-optimal N can cause yield losses and increases the potential for leaching loss during the following winter (Islam and Evans, 1994; Shepherd and Sylvester-Bradley 1996; Sieling et al., 1997; Engström et al., 2009).

N uptake by the seeds ranged from 69 kg/N ha in an unfertilized control to 175 kg/N ha in plots which had received 320 kg N/ha (Aniol, 1993), thus leaving large amounts of fertilizer N in the system. Crop recovery of labeled fertilizer applied to oilseed rape in spring varied considerably with the soil and year, ranging from 31 to 70% (Chalmers and Darby, 1992; Jensen et al., 1997; Macdonald et al., 1997; Sieling et al., 1998a; Beims, 2005). Approximately half of the fertilizer-derived nitrogen was contained in the seed, and, where recovery of labeled nitrogen by the crop was poor, up to 39% of labeled fertilizer remained in the soil.

In Germany N fertilizers are commonly applied according to growth stage (Rathke et al., 2006), taking soil mineral N content at the start of spring growth into account by the 'Nmin method' developed by Wehrmann and Scharpf (1979). In general, high canopy N is considered in the calculation of N fertilization rates by subtracting a fixed value of 20 kg N/ha, which hardly accounts for the actual canopy development, and corresponding N amounts taken up by the canopy before fertilization. In contrast, balance-sheet methods regarding the amounts of mineralization of soil-borne N and the amount of N in rapeseed canopies at the end of fall and the end of winter have been developed in France (Reau et al., 1994; Makowski et al., 2005), resulting in an optimized N fertilization of oilseed rape with on average lower doses of N fertilizer (Hébinger, 2009).

In a series of field trials consisting of seven sites and two years, Henke et al. (2009) tried to verify this approach under different climatic conditions in order to derive site-specific N fertilization rates. First, several methods which can easily be used by the farmers for estimating canopy N were tested. The use of digital photos giving information on the leaf area index, and in consequence on the above-ground biomass and N uptake, was limited to canopies with a leaf area index <1. Also counting the number of leaves was not suitable for estimating the amount of N in the aerial biomass, whereas weighing above-ground fresh matter, as done in France, gave good results. Estimating canopy N by weighing fresh matter of a defined area with several replications is based on the assumption that aerial dry matter content and N concentrations vary only a little at early growth stages. Reau et al. (1994) found a nearly constant N concentration of 4.5% N (in dry matter) at early growth stages of oilseed rape. Colnenne et al. (1998) described a critical N dilution curve for winter oilseed rape with a constant N concentration of 4.48% at early growing stages. N concentration decreased when dry matter (t/ha) exceeded 1 t dry matter/ha following the equation  $N_{conc} = 4.48 \text{ W}^{-0.25}$ , where Nconc is the N concentration in the aerial dry matter (% N) and W the shoot dry matter (t/ha). The N concentration of 4.5% found by Henke (2008) in aerial dry matter confirmed the results of Reau et al. (1994) and Colnenne et al. (1998). The measured dry matter contents in fall averaged 10%, leading to conversion factors of 45 in fall (kg N/ha in the canopy = 45 \* kg fresh matter/ $m^2$ ) (Henke, 2008).

In a second step, economically optimal N fertilizer amounts were estimated from quadratic N response curves and related to the soil mineral N content in spring, and N in the rapeseed canopy in fall and/or in spring. The site caused a large variation; however, no interactions with the parameters were significant, meaning that the effects were similar at all sites. Soil mineral N in spring showed no correlation with the optimal N fertilizer amounts, since oilseed rape can take up large amounts of N before winter and therefore often depletes the mineral N pool in the soil. Even canopy N in spring only slightly correlated with the optimal N fertilizer amounts, since leaf losses, and consequently N losses, over winter highly influence canopy N at the beginning of spring growth (Dejoux et al., 2003). Based on 2 years, measuring canopy N in fall gave the best results. The regression (Fig. 3) showed a significant negative relationship between canopy N in fall and the optimal N fertilizer amounts. The slope of this regression was -0.7, indicating that these N amounts should be partly considered when calculating N fertilization rates (Henke et al., 2009).

Experimental results on the impact of canopy N in fall or at the beginning of spring growth on the optimal



Fig. 3 Regression of the optimal N fertilization on canopy N in fall for seven experimental sites in Germany in two years (Henke et al., 2009)

N fertilizer amounts are scarce. Gabrielle et al. (1998) and Dejoux et al. (1999a, b) reported N losses caused by leaf losses due to frost over winter. Dejoux et al. (1999a) found a recovery of the N losses over winter by the rapeseed crop in spring of about 40% under the climatic conditions of their experimental site located in the Paris Basin (France). This recovery could be explained by the synchronization of lost leaves' decomposition and N uptake by the rapeseed crop in spring (Dejoux et al., 2003). If leaf losses occur over winter, N from dropped leaves is quickly mineralized in spring. Dejoux et al. (1999a) measured a decomposition fraction of added <sup>15</sup>N-labeled leaves of 0.94 before harvest. However, Dejoux et al. (2003) also found a fraction of gaseous N losses of 0.4 of N applied with frozen leaves estimated by unrecovered <sup>15</sup>N. These results indicate that N losses over winter are likely to be recovered in spring to some extent, and consequently N fertilization rates can be reduced. This recovery potential is totally neglected when canopy N in spring is used as an indicator for N fertilization rates in spring. Additionally, Dejoux et al. (2003) reported that the differences between canopies which varied largely in fall decreased during winter. Consequently, N fertilization rates would be overestimated if canopy N in spring was taken into account when deriving N fertilization rates. For these reasons canopy N in fall seems to be a more favorable indicator compared with canopy N in spring for the derivation of N fertilization

rates. Besides the recovery of N derived from fallen leaves in spring, N uptake of unfertilized oilseed rape canopies in fall could be an indicator for the N mineralization potential of the location. In the case of insignificant leaf losses over winter, a well-developed rapeseed canopy showed a higher leaf area index at the beginning of spring growth than a poor canopy, which, in consequence, allows for reduced N fertilization rates (Mendham et al., 1981).

The results of Henke et al. (2009) clearly demonstrate that canopy N in fall is a suitable indicator for optimizing N fertilization rates. The implementation of this approach into practical use by farmers is based on an average rapeseed canopy with 50 kg N/ha accumulated in fall  $(1.1 \text{ kg FM}/\text{m}^2)$  which will be fertilized in spring according to the official recommendation (e.g. 200 kg N/ha). Each kg N/ha exceeding the threshold of 50 kg N/ha reduces the N fertilization in spring by 0.7 kg N/ha. However, it is not possible to give absolute recommendations. In addition, the causes of the large site-to-site variations remain to be clarified. On the other hand, besides a reduction of N fertilization if canopy N is high, the study does not indicate that N fertilizer rates need to be increased if canopy N in fall is very poor because the crop is often not able to utilize the additionally applied N for its yield formation.

In the future the algorithm developed by Henke et al. (2009) allows optimizing N fertilization of oilseed rape site-specifically. Results of Müller et al. (2008) and Behrens and Diepenbrock (2006) showed that N uptake of oilseed rape in fall can be estimated on a field scale using vegetation indices derived from hyperspectral reflection measurements. Based on these data, an application map for the N fertilization in spring can be generated which takes the withinfield variation into account according to the approach presented by Henke et al. (2009). In France, a system based on satellite observations called 'Farmstarcolza' is currently being tested to draw a map of the fields with nitrogen supply advice, to practice modular application on the crops and to increase the precision of the crop N absorption assessment (Lagarde and Champolivier, 2006; Hébinger, 2009).

### 4 N Dynamic After Oilseed Rape

In contrast to the topics discussed above, soil tillage following oilseed rape does not affect the N balance of the crop, but strongly influences the N dynamics in the soil. As shown above, rapeseed demands high levels of nitrogen fertilizer, often exceeding 200 kg N/ha to achieve maximum yields. However, N offtake by the seed and N harvest index, ranging between 0.6 and 0.7, are low compared with cereals (Shepherd and Sylvester-Bradley, 1996; Sieling et al., 1999; Malagoli et al., 2005). Although rapeseed residues lead to N immobilization after incorporation into the soil (Jensen et al., 1997; Justes et al., 1999; Trinsoutrot et al., 2000), soil mineral N content increases regularly during fall (Lickfett, 1993). In consequence, large amounts of nitrate are likely to be leached with drainage water during the subsequent percolation period (Sieling et al., 1999; Sieling and Kage, 2006).

Increasing amounts of N fertilization to oilseed rape strongly raised soil mineral N content at the harvest of oilseed rape and during the subsequent fall. Several studies resulted in a remarkably higher soil mineral N content, sometimes exceeding 100 kg soil mineral N/ha at oilseed rape harvest following a two-straightline function if high amounts of N fertilizer were used (Shepherd and Sylvester-Bradley, 1996; Sieling et al., 1999; Beaudoin et al., 2005; Makowski et al., 2005). Di and Cameron (2002) reported a threshold level of N fertilization above 200 kg N/ha which increased N leaching in arable cropping. Sieling and Kage (2006) stated a positive correlation between a simple N balance, calculated from N fertilization minus N offtake by the seed, and N leaching, and between mineral N fertilization and N leaching. This positive correlation between excessive N fertilization and N leaching is also confirmed by the scenario calculations made by Henke et al. (2008). Amounts of N fertilizer above 200 kg N/ha led to a strong increase in N leaching, whereas amounts below 200 kg N/ha differed slightly. These results show the importance of more precise calculations of N fertilizer doses considering soil and oilseed rape canopy properties (see Chap. 3).

According to Di and Cameron (2002), the prevention of soil mineral N accumulation in the soil after harvest is the key to reducing N leaching. There are several well-known agronomic measures to reduce mineralization of soil organic N, and in consequence, nitrate leaching after harvest. First of all, reducing tillage depth and delaying tillage after harvest diminish soil disturbance (Goss et al., 1993), and consequently soil N release, since intensive tillage operations after harvest stimulate net mineralization of soil-borne N due to soil disturbance (Lickfett, 1993). Using a combined approach of field trials and modeling, Henke et al. (2008) observed that minimum tillage combined with a short period of growing volunteer rapeseed as a catch crop before sowing of winter wheat can decrease N leaching.

Due to the tillage operations, however, not only the soil is mixed but also oilseed rape residues are incorporated into the soil. The C/N ratio in oilseed rape residues under a regular fertilization regime of about 200 kg N/ha was about 50 to 60. Justes et al. (1999) and Mary et al. (1999) report N immobilization of about 20 kg N/ha after rapeseed residue incorporation (C/N = 54, N fertilization 270 kg N/ha). Jensen et al. (1997) observed a decrease in soil mineral N of 18 and 25 kg N/ha after incorporation of 4 or 8 t oilseed rape residues/ha, respectively. In addition, Trinsoutrot et al. (2000) stated that the N amounts mineralized from rapeseed residues and then returned to the subsequent crop are relatively small. The studies showed that straw incorporation is an indispensable part of proper N management after harvest to prevent N leaching (Justes et al., 1999; Trinsoutrot et al., 2000; Beaudoin et al., 2005).

Secondly, changes in the crop rotation such as the introduction of catch crops and spring crops can decrease soil mineral N content during fall and therefore reduce the risk of nitrate leaching after oilseed

rape. However, growing catch crops combined with a spring crop is less profitable than cropping winter wheat, which is commonly grown after oilseed rape, and therefore compensatory payments would be necessary, especially in water protection areas. Justes et al. (1999) reported a significant reduction of the soil mineral N content and nitrate leaching by a radish cover crop or volunteer rapeseed compared with a bare soil. Henke et al. (2008) suggested Phacelia as a catch crop, whereas Lickfett (1993) recommended volunteer oilseed rape to grow over winter without any tillage operation, thus reducing mineralization of soil-borne N, and consequently N leaching. However, additionally growing volunteer rapeseed is characterized by two disadvantages. Firstly, due to a lower drainage rate, the nitrate concentration in the percolation water can rise remarkably and exceed the threshold for drinking water of the European Union to a higher extent than the other treatments (Henke et al., 2008). Secondly, growing volunteer rapeseed could be critical because a high density of oilseed rape in the rotation can promote pathogens and pests (Christen, 2006).

### 5 Conclusion

Concerning N leaching, oilseed rape is an ambivalent crop. In fall, oilseed rape can take up large amounts of N and prevent it from leaching, whereas after harvest, due to a low N-use efficiency, N surplus and, in consequence, N losses via leaching are often high, especially if wheat follows. In order to minimize the environmental impact, farmers have to reconsider critically N fertilization of oilseed rape and to skip practices that have only small effects on yield, e.g. N fertilization in fall. In addition, N already taken up by the crop before winter should be taken into account when estimating spring N fertilization. Each kg fall canopy N exceeding 50 kg N/ha reduces spring N fertilization by 0.7 kg N/ha under German conditions. However, under the currently high prices for oilseed rape, some management measures might make sense from the economic point of view, being an important aspect for the farmers, although they do not decrease the N surpluses. Apart from the discussion of the N balance, appropriate management after oilseed rape harvest can reduce N leaching, if minimum tillage (or even no-till) is used and/or catch crops are grown.

Despite the problems addressed above, it should be pointed out that oilseed rape is indispensable as a favorable preceding crop for cereals, especially wheat. Growing legumes will raise similar problems as in oilseed rape, whereas the area of sugar beets is limited. Increasing the percentage of wheat within the rotation, up to monoculture, is also not sustainable, because wheat yield decreases while the amount of fertilizer N will increase to achieve optimal yields.

### References

- Allison M.F., Armstrong M.J., Jaggard K.W., Todd A.D. (1998) Integration of nitrate cover crops into sugar beet (*Beta vul-garis*) rotations. I. Management and effectiveness of nitrate cover crops, J. Agr. Sci. 130, 53–60.
- Aniol H. (1993) Ertrag, Ertragsstruktur und Stickstoffaufnahme von Winterraps (*Brassica napus* L.) in Abhängigkeit von N-Menge, N-Verteilung und Genotyp, Ph.D. thesis, University Kiel.
- Aufhammer W., Kübler E., Bury M. (1994) Stickstoffaufnahme und Stickstoffrückstände von Hauptfrucht- und Ausfallrapsbeständen, J. Agron. Crop Sci. 172, 255–264.
- Barraclough P.B. (1989) Root growth, macro-nutrient uptake dynamics and soil fertility requirements of a high-yielding winter oilseed rape crop, Plant Soil 119, 59–70.
- Beaudoin N., Saad J.K., van Laethem C., Machet J.M., Maucorps J., Mary B. (2005) Nitrate leaching in intensive agriculture in Northern France: Effect of farming practices, soils and crop rotations, Agr. Ecosys. Environ. 111, 292–310.
- Behrens T., Diepenbrock W. (2006) Using hemispherical radiation measurements to predict weight-related growth traits in oilseed rape (*Brassica napus* L.) and barley (*Hordeum vulgare* L.) canopies, J. Agron. Crop Sci. 192, 465–474.
- Beims S. (2005) Untersuchungen zur N-Effizienz und zum N-Mineralisationspotenzial in langjährig unterschiedlichen Düngungssystemen mit Hilfe von <sup>15</sup>N-markiertem Mineraldünger, Schriftenreihe des Instituts für Pflanzenbau und Pflanzenzüchtung der Agrarwissenschaftlichen Fakultät der CAU, Bd. 38.
- Berry P. (2009) Breeding for improved nitrogen use efficiency in oilseed rape, European Workshop 'N fertilization of winter oilseed rape', 23–24 March 2009, Berlin, Germany, http:// www.pflanzenbau.uni-kiel.de/veranstaltungen/n-duengung\_ wraps/Berry.pdf (08/09/2009).
- Bilsborrow P.E., Evans E.J., Zhao F.J. (1993) The influence of spring nitrogen on yield, yield components and glucosinolate content of autumn-sown oilseed rape (*Brassica napus*), J. Agr. Sci. 120, 219–224.
- Boelcke B., Léon J., Schulz R.R., Schröder G., Diepenbrock W. (1991) Yield stability of winter oilseed rape (*Brassica napus* L.) as affected by stand establishment and nitrogen fertilization, J. Agron. Crop Sci. 167, 241–248.
- Chalmers A.G. (1989) Autumn and spring fertiliser nitrogen requirements for winter oilseed rape, Asp. Appl. Biol. 23, 125–133.

- Chalmers A.G., Darby R.J. (1992) Nitrogen application to oilseed rape and implications for potential leaching loss, Asp. Appl. Biol. 30, 425–430.
- Christen O. (2006) Raps in der Fruchtfolge ist die Grenze der Ausdehnung erreicht? (with English summary), UFOP-Schriften 29, Öl- und Proteinpflanzen, OIL 2005, pp. 119–128.
- Colnenne C., Meynard J.-M., Reau R., Justes E., Merrien A. (1998) Determination of a critical nitrogen dilution curve for winter oilseed rape, Ann. Bot. 81, 311–317.
- Colnenne C., Meynard J.-M., Roche R., Reau R. (2002) Effects of nitrogen deficiencies on autumnal growth of oilseed rape, Eur. J. Agron. 17, 11–28.
- Davies D.B., Sylvester-Bradley R. (1995) The contribution of fertiliser nitrogen to leachable nitrogen in the UK: A review, J. Sci. Food Agr. 68, 399–406.
- Darby R.J., Hewitt M.V. (1990) A comparison of the effects of single or multiple spring applications of prilled urea or 'Nitro-chalk' to winter oilseed rape (*Brassica napus*), J. Agr. Sci. 115, 363–368.
- Dejoux J.-F., Recous S., Meynard J.-M, Trinsoutrot I., Leterme P. (1999a) The fate of nitrogen from winterfrozen rapeseed leaves: mineralization, fluxes to the environment and uptake by rapeseed crop in spring, Plant Soil 218, 257–272.
- Dejoux J.-F., Meynard J.-M., Reau R. (1999b) Rapeseed new crop management with early sowing in order to reduce N-leaching and N-fertilization, in: New Horizons for an Old Crop, Proceedings of the 10th International Rapeseed Congress, GCIRC: CD-ROM, Canberra, Australia.
- Dejoux J.-F., Meynard J.-M., Reau R., Roche R., Saulas P. (2003) Evaluation of environmentally-friendly crop management systems based on very early sowing dates for winter oilseed rape in France, Agronomie 23, 725–736.
- Di H.J., Cameron K.C. (2002) Nitrate leaching in temperate agroecosystems: sources, factors and mitigation strategies, Nutr. Cycl. Agroecosys. 46, 237–256.
- Diepenbrock W. (2000) Yield analysis of winter oilseed rape (Brassica napus L.): a review, Field Crop. Res. 67, 35–49.
- Doluschitz R., Welck H., Zeddies J. (1992) Stickstoffbilanzen landwirtschaftlicher Betriebe - Einstieg in die ökologische Buchführung? Ber. Landwirtsch. 70, 551–565.
- Düngeverordnung (2006) Bekanntmachung der Neufassung der Düngeverordnung, Bundesgesetzblatt I 2, 33–43.
- Engels T., Kuhlmann H. (1993) Effect of the rate of fertilizer on apparent net mineralization of N during and after cultivation of cereal and sugar beet crops, Z. Pflanzenernachr. Bodenk. 154, 149–154.
- Engström L., Stenberg M., Lindén B. (2009) Methods to reduce nitrate leaching after winter oilseed rape and peas, European Workshop 'N fertilization of winter oilseed rape', 23–24 March 2009, Berlin, Germany, http://www.pflanzenbau. uni-kiel.de/veranstaltungen/n-duengung\_wraps/Engstroem. pdf (08/09/2009).
- European Union (1991) Implementation of nitrate Directive (Directive 91/676/EEC), URL: http://ec.europa.eu/environment/ water/water-nitrates/directiv.html (31/10/2006).
- European Union (2000) The EU Water Framework Directive integrated river basin management for Europe, URL: http:// ec.europa.eu/environment/water/water-framework/index\_ en.html (07/03/2008).

- Flénet F., Sieling K., Wagner D., Colnenne C. (2009) Is there a need for N fertilizer in autumn in winter oilseed rape (WOSR)? European Workshop 'N fertilization of winter oilseed rape', 23–24 March 2009, Berlin, Germany, http://www.pflanzenbau.uni-kiel.de/veranstaltungen/ n-duengung wraps/Flenet.pdf (08/09/2009).
- Fraters D., Boumans L.J.M., Van Drecht G., De Haan T., De Hoop W.D. (1998) Nitrogen monitoring in groundwater in the sandy regions of the Netherlands, Environ. Pollut. 102, 479–485.
- Gabrielle B., Denoroy P., Gosse G., Justes E., Anderson M.N. (1998) A model of leaf area development and senescence for oilseed rape, Field Crop. Res. 57, 209–222.
- Glendining M.J., Powlson D.S., Poulton P.R., Bradbury N.J., Palazzo D., Li X. (1996) The effects of long-term applications of inorganic nitrogen fertilizer on soil nitrogen in the Broadbalk Wheat Experiment, J. Agr. Sci. 127, 347–363.
- Goss M.J., Howse K.R., Lane P.W., Christian D.G., Harris G.L. (1993) Losses of nitrate-nitrogen in water draining from under autumn-sown crops established by direct drilling or mouldboard ploughing, J. Soil Sci. 44, 35–48.
- Goss M.J., Williams B.L., Howse K.R. (1991) Organic matter turnover and nitrate leaching, in: Wilson W.S. (Ed.), The advances in soil organic matter research: The impact on agriculture and the environment, The Royal Society of Chemistry, pp. 107–114.
- Goulding K.W.T., Poulton P.R., Webster C.P., Howe M.T. (2000) Nitrate leaching from the Broadbalk Wheat Experiment, Rothamsted, UK, as influenced by fertilizer and manure inputs and the weather, Soil Use Manage. 16, 244–250.
- Grosse F., Léon J., Diepenbrock W. (1992) Untersuchungen zur Ertragsbildung und Ertragsstruktur bei Winterrps (*Brassica napus* L.). I. Genotypische Variabilität, J. Agron. Crop Sci. 169, 70–93.
- Hébinger H. (2009) N fertilization decision support systems for WOSR based on a predictive balance sheet method, European Workshop 'N fertilization of winter oilseed rape', 23–24 March 2009, Berlin, Germany, http://www. pflanzenbau.uni-kiel.de/veranstaltungen/n-duengung\_ wraps/Hebinger.pdf (08/09/2009).
- Henke J. (2008) Entwicklung und Bewertung von Strategien zur Verbesserung der Stickstoffeffizienz im Winterrapsanbau, Schriftenreihe des Instituts für Pflanzenbau und Pflanzenzüchtung der Agrarwissenschaftlichen Fakultät der CAU, Band 54.
- Henke J., Böttcher U., Neukam D., Sieling K., Kage K. (2008) Evaluation of different agronomic strategies to reduce nitrate leaching after winter oilseed rape (*Brassica napus* L.) using a simulation model, Nutr. Cycl. Agoecosys. 82, 299–314.
- Henke J., Sieling K., Kage H., Breustedt G. (2007) Impact of uncertainty on the optimum nitrogen fertilization rate and agronomic, ecological and economic factors in an oilseed rape based crop rotation, J. Agr. Sci. 145, 455–468.
- Henke J., Sieling K., Sauermann W., Kage H. (2009) Analysing soil and canopy factors affecting optimum nitrogen fertilization rates of oilseed rape (*Brassica napus*), J. Agr. Sci. 147, 1–8.
- Hocking P.J., Randall P.J., DeMarco D. (1997) The response of dryland canola to nitrogen fertilizer: partitioning and mobilization of dry matter and nitrogen, and nitrogen effects on yield components, Field Crop. Res. 54, 201–220.

- Islam N., Evans E.J. (1994) Influence of lodging and nitrogen rate on yield and yield attributes of oilseed rape (*Brassica napus* L.), Theor. Appl. Genet. 88, 530–534.
- Jansons V., Buismanis P., Dzalbe I., Kirsteina D. (2003) Catchment and drainage field nitrogen balances and nitrogen loss in three agriculturally influenced Latvian watersheds, Eur. J. Agron. 20, 173–179.
- Jensen L.S., Christensen L., Mueller T., Nielsen N.E. (1997) Turnover of residual <sup>15</sup>N-labelled fertilizer N in soil following harvest of oilseed rape (*Brassica napus* L.), Plant Soil 190, 193–202.
- Justes E., Mary B., Nicolardot B. (1999) Comparing the effectiveness of radish cover crop, oilseed rape volunteers and oilseed rape residues incorporation for reducing nitrate leaching, Nutr. Cycl. Agroecosys. 55, 207–220.
- Kelm M., Neumann H., Hüwing H., Kemper N. (2007) Wissen wo man steht. Landwirtschaftliche Produktionssysteme in Schleswig-Holstein: Leistungen und ökologische Effekte. Ergebnisse des Projektes COMPASS, Institut für Pflanzenbau und Pflanzenzüchtung -Grünland und Futterbau/ Ökologischer Landbau- und Institut für Phytopathologie der CAU Kiel, ISBN 3-9811625-0-9.
- Kuhlmann H., Engels T. (1989) Nitrogen utilisation in relation to N-fertilisation, The Fertiliser Society of London 287, 28–43.
- Lagarde F., Champolivier L. (2006) Le raisonnement et les avancées techniques permettent de réduire la fertilization azotée : le cas de Farmstar-colza<sup>®</sup>, Olégineux Crops, Gras Lipides 13, 384–387.
- Lickfett T. (1993) Auswirkungen verminderter Produktionsintensität in zwei Rapsfruchtfolgen auf Elemente des N-Haushaltes im System Boden-Pflanze, PhD thesis, University of Göttingen.
- Lindén B., Walgren B. (1993) Nitrogen mineralization after leys ploughed in early or late autumn, Swed. J. Agr. Res. 23, 77–89.
- Lord E.I., Anthony S.G., Goodlass G. (2002) Agricultural nitrogen balance and water quality in the UK, Soil Use Manage. 18, 363–369.
- Macdonald A.J., Powlson D.S., Poulton P.R., Jenkinson D.S. (1989) Unused fertiliser nitrogen in arable soils - its contribution to nitrate leaching, J. Sci. Food Agr. 46, 407–419.
- Macdonald A.J., Poulton P.R., Powlson D.S., Jenkinson D.S. (1997) Effects of season, soil type and cropping on recoveries, residues and losses of <sup>15</sup>N-labelled fertilizer applied to arable crops in spring, J. Agr. Sci. 129, 125–154.
- Makowski D., Maltas A., Morison M., Reau R. (2005) Calculating N fertilizer doses for oil-seed rape using plant and soil data, Agron. Sustain. Dev. 25, 159–161.
- Malagoli P., Laine P., Rossato L., Ourry A. (2005) Dynamics of nitrogen uptake and mobilization in field grown winter oilseed rape (*Brassica napus*) from stem extension to harvest. 1. Global N flows between vegetative and reproductive tissues in relation to leaf fall and their residual N, Ann. Bot. 95, 853–861.
- Mary B., Beaudoin N., Justes E., Machet J.M. (1999) Calculation of nitrogen mineralization and leaching in fallow soil using a simple dynamic model, Eur. J. Soil Sci. 50, 549–566.
- Mendham N.J., Shipway P.A., Scott R.K. (1981) The effects of delayed sowing and weather on growth, development and yield of winter oil-seed rape (*Brassica napus* L.), J. Agr. Sci. 96, 389–416.

- Merrien A., Palleau J.P., Maisonneuve C. (1988) Besoins en éléments minéraux du colza cultivé en France, in: Physiologie et élaboration du rendement du Colza, Paris: Cétiom Editions, pp. 34–46.
- Müller K., Böttcher U., Meyer-Schatz F., Kage H. (2008) Analysis of vegetation indices derived from hyperspectral reflection measurements for estimating crop canopy parameters of oilseed rape (*Brassica napus* L.), Biosyst. Eng. 101, 172– 182.
- Öborn I., Edwards A.C., Witter E., Oenema O., Ivarsson K., Withers P.J.A., Nilsson S.I., Richert Stinzing A. (2003) Element balances as a tool for sustainable nutrient management: a critical appraisal of their merits and limitations within an agronomic and environmental context, Eur. J. Agron. 20, 211–225.
- Ogilvy S.E., Bastiman B. (1992) The effect of rate and timing of autumn nitrogen on the pre-flowering dry matter production and seed yield of winter oilseed rape, Asp. Appl. Biol. 30, 413–416.
- Rathke G.W., Behrens T., Diepenbrock W. (2006) Integrated nitrogen management strategies to improve seed yield, oil content and nitrogen efficiency of winter oilseed rape (*Brassica napus* L.): A review, Agr. Ecosyst. Environ. 117, 80–108.
- Rathke G.-W., Christen O., Diepenbrock W. (2005) Effects of nitrogen source and rate on productivity and quality of winter oilseed rape (*Brassica napus* L.) grown in different crop rotations, Field Crop. Res. 94, 103–113.
- Reau R., Wagner D., Palleau J.P. (1994) End of winter diagnosis: Winter rapeseed and nitrogen fertilization, in: Borin M., Sattin M. (Eds.), Proceedings of the third congress of the European Society for Agronomy, Podova: European Society for Agronomy, pp. 220–221.
- Rinnofner T., Friedel J.K., de Kruijff R., Pietsch G., Freyer B. (2008) Effect of catch crops on N dynamics and following crops in organic farming, Agron. Sustain. Dev. 28, 551–558.
- Rossato L., Lainé P., Ourry A. (2001) Nitrogen storage and remobilisation in *Brassica napus* L. during the growth cycle: nitrogen fluxes within the plant and changes in soluble protein patterns, J. Exp. Bot. 52, 1655–1663.
- Sacco D., Bassanimo M., Grignani C. (2003) Developing a regional agronomic information system for estimating nutrient balances at a larger scale, Eur. J. Agron. 20, 199–210.
- Schjoerring J.K., Bock J.G.H., Gammelvind L., Jensen C.R., Mogensen, V.O. (1995) Nitrogen incorporation and remobilisation in different shoot components in field-grown winter oilseed rape (*Brassica napus* L.) as affected by rate of nitrogen application and irrigation, Plant Soil 177, 255–264.
- Shepherd M.A., Lord E.I. (1996) Nitrate leaching from a sandy soil: the effect of previous crop and post-harvest soil management in an arable rotation, J. Agr. Sci. 127, 215–229.
- Shepherd M.A., Sylvester-Bradley R. (1996) Effect of nitrogen fertilizer applied to winter oilseed rape (*Brassica napus*) on soil mineral nitrogen after harvest and on the response of a succeeding crop of winter wheat to nitrogen fertilizer, J. Agr. Sci. 126, 63–74.
- Sieling K. (2005) Growth stage specific application of slurry and mineral N to oilseed rape, wheat and barley, J. Agr. Sci. 142, 495–502.

- Sieling K., Hanus H. (1997) N-Aufnahme und N-Verwertungseffizienz zweier Weizensorten bei variierter mineralischer N-Düngung, Pflanzenbauwissenschaften 1, 57–62.
- Sieling K., Kage H. (2006) N balance as an indicator of N leaching in an oilseed rape - winter wheat - winter barley rotation, Agr. Ecosyst. Environ. 115, 261–269.
- Sieling K., Kage H. (2007) Autumnal N fertilization of late sown oilseed rape after minimum tillage, Proceedings of the 12th International Rapeseed Congress (Wuhan, China), III (Agronomy), pp. 375–378.
- Sieling K., Günther-Borstel O., Hanus H. (1997) Effects of slurry application and mineral nitrogen fertilization on N leaching in different crop combinations, J. Agr. Sci. 128, 79–86.
- Sieling Günther-Borstel O., Teebken T., Hanus H. (1999) Soil mineral N and N net mineralization during autumn and winter under an oilseed rape – winter wheat – winter barley rotation in different crop management systems, J. Agr. Sci. 132, 127–137.
- Sieling K., Schröder H., Hanus H. (1998a) Mineral and slurry nitrogen effects on yield, N uptake, and apparent Nuse efficiency of oilseed rape (*Brassica napus*), J. Agr. Sci. 130, 165–172.
- Sieling K., Schröder H., Hanus H. (1998b) Yield, N uptake, and apparent N-use efficiency of winter wheat and winter barley grown in different cropping systems, J. Agr. Sci. 131, 375–387.
- Smith K.A., Chambers B.J. (1992) Improved utilisation of slurry nitrogen for arable cropping, Asp. Appl. Biol. 30, 127–134.

- Smith S.J., Schepers S.J., Porter L.K. (1990) Assessing and managing agricultural nitrogen losses to environment, Adv. Soil Sci. 14, 1–13.
- Spink J. (2009) Nitrogen rate and timing to achieve an optimum canopy size, European Workshop 'N fertilization of winter oilseed rape', 23–24 March 2009, Berlin, Germany, http:// www.pflanzenbau.uni-kiel.de/veranstaltungen/n-duengung\_ wraps/Spink.pdf (08/09/2009).
- Sylvester-Bradley R., Chambers B.J. (1992) The implications of restricting use of fertilizer nitrogen for the productivity of arable crops, their profitability and potential pollution by nitrate, Asp. Appl. Biol. 30, 85–94.
- Trinsoutrot I., Nicolardot B., Justes E., Recous S. (2000) Decomposition in the field of residues of oilseed rape grown at two levels of nitrogen fertilisation. Effects on the dynamics of soil mineral nitrogen between successive crops, Nutr. Cycl. Agroecosys. 56, 125–137.
- Webb J., Harrison R., Ellis S. (2000) Nitrogen fluxes in three arable soils in the UK, Eur. J. Agron. 13, 2007–223.
- Wehrmann J., Scharpf H.C. (1979) Der Mineralstickstoffgehalt des Bodens als Maßstab für den Stickstoffdüngerbedarf (Nmin-Methode), Plant Soil 52, 109–126.
- Zerulla W., Isermann K., Henjes G. (1993) Einfluss langjähriger unterschiedlicher N-Düngung auf Ertrag und N-Verlagerung im Bodenprofil verschiedener Ackerböden, Agribiol. Res. 46, 277–285.
- Zhao F.J., Evans E.J., Bilsborrow P.E., Syers J.K. (1993) Influence of sulphur and nitrogen on seed yield and quality of low glucosinolate oilseed rape (*Brassica napus* L.), J. Sci. Food Agr. 63, 29–37.
# Improving Nitrogen Fertilization in Rice by Site-Specific N Management\*

Shaobing Peng, Roland J. Buresh, Jianliang Huang, Xuhua Zhong, Yingbin Zou, Jianchang Yang, Guanghuo Wang, Yuanying Liu, Ruifa Hu, Qiyuan Tang, Kehui Cui, Fusuo Zhang, and Achim Dobermann

# Contents

1	Introduction	944
2	Principles and Procedures of Site-Specific N Management (SSNM)	945
3	IRRI-China Collaboration on SSNM Research	946
4	Key Research Findings	946
5	Remarks on Site-Specific N Management	950
6	Conclusions	951
Re	eferences	951

S. Peng (🖂)

Abstract Excessive nitrogen (N) application to rice (Oryza sativa L.) crop in China causes environmental pollution, increases the cost of rice farming, reduces grain yield and contributes to global warming. Scientists from the International Rice Research Institute have collaborated with partners in China to improve rice N fertilization through site-specific N management (SSNM) in China since 1997. Field experiments and demonstration trials were conducted initially in Zhejiang province and gradually expanded to Guangdong, Hunan, Jiangsu, Hubei and Heilongjiang provinces. On average, SSNM reduced N fertilizer by 32% and increased grain yield by 5% compared with farmers' N practices. The yield increase was associated with the reduction in insect and disease damage and improved lodging resistance of rice crop under the optimal N inputs. The main reason for poor fertilizer N use efficiency of rice crop in China is that most rice farmers apply too much N fertilizer, especially at the early vegetative stage. We observed about 50% higher indigenous N supply capacity in irrigated rice fields in China than in other major rice-growing countries. Furthermore, yield response of rice crop to N fertilizer application is low in China, around  $1.5 \text{ t ha}^{-1}$  on average. However, these factors were not considered by rice researchers and extension technicians in determining the N fertilizer rate for recommendation to rice farmers in China. After a decade of research on SSNM in China and other Asian rice-growing countries, we believe SSNM is a matured technology for improving both fertilizer N use efficiency and grain yield of rice crop. Our challenges are to further simplify the procedure of SSNM and to convince policy-makers of the effectiveness of this technology in order to facilitate a wider adoption of SSNM among rice farmers in China.

<sup>\*</sup> Article from the International Workshop on Sustainable Nutrient Management: Technology and Policy organized by Prof. Fusuo Zhang and Dr. Yuxin Miao, May 28–21, 2008, Shijiazhuang, Hebei Province, China.

Crop and Environmental Sciences Division, International Rice Research Institute, DAPO Box 7777, Metro Manila, Philippines e-mail: s.peng@cgiar.org

**Keywords** Site-Specific nitrogen management • Nitrogen use efficiency • Grain yield • Nitrogen response•Rice•China

# 1 Introduction

Increase in fertilizer nutrient input, especially N fertilizer, has contributed significantly to the improvement of crop yields in the world (Cassman et al., 2003). Development of semi-dwarf rice (*Oryza sativa* L.) varieties in the 1960s assured the achievement of high grain yield under increased N fertilizer rates because of their lodging resistance at high N inputs (Yoshida, 1972). To maximize grain yield, farmers often apply a higher amount of N fertilizer than the minimum required for maximum crop growth (Lemaire and Gastal, 1997). Overapplication of N fertilizer may actually decrease grain yield by increasing susceptibility to lodging (Pham et al., 2004) and damage from pests and diseases (Cu et al., 1996).

China is currently the world's largest consumer of N fertilizer. In 2006, annual N fertilizer consumption in China was 31 million metric tons or 31.7% of the global N consumption (Heffer, 2008). About 18% of this N was used for rice production in China (Heffer, 2008). China accounts for about 19% of the world's rice planting area and produced 29% of global rice production in 2006 (FAOSTAT, 2008). Rice crops in China use about 36% of the total N fertilizer used for rice production in the world (Heffer, 2008).

There are several sources of information on the average N rate of rice production in China. The International Fertilizer Industry Association (IFA, 2002) reported that China's national average N rate for rice was 145 kg ha<sup>-1</sup> in 1997. Based on data from 1995 to 1997, the average rate of N application for rice production in China was 180 kg ha<sup>-1</sup> (FAO, unpublished data, 2001). If we use the data for the rice planting area from FAOSTAT and N consumption for rice from Heffer (2008), China's national average N rate for rice was 193 kg ha<sup>-1</sup> in 2006, about 90% higher than the world average (FAOSTAT, 2008; Heffer, 2008). Nitrogen rates of 150 to 250 kg ha<sup>-1</sup> are common (Wang et al., 2001; Peng et al., 2006). In Jiangsu Province, China, the average N rate reached  $300 \text{ kg ha}^{-1}$  in some counties (Q. Zhu, pers. commun., 2001). Average rice grain yield in China was 6.27 t ha<sup>-1</sup> in 2006, about 53% higher than the world average (FAOSTAT, 2008). Rice crop in China received twice as much N fertilizer as in Japan but the grain yield was similar in the two countries.

Fertilizer N use efficiency can be measured in different ways using N-omission plots (Novoa and Loomis, 1981). The yield increase that results from N application in comparison with no N application is defined as the agronomic N use efficiency (kg grain yield increase per kg N applied). The apparent recovery efficiency of fertilizer N is used to express the percentage of fertilizer N recovered in aboveground plant biomass at the end of the cropping season. Yield response of rice crop to N fertilizer application is defined as N response, which is calculated as the difference between yields with and without N fertilizer (IRRI, 2006). The partial factor productivity from applied N is the ratio of grain yield to N applied and it provides an integrative index that quantifies total economic output relative to utilization of all N resources in the system, including indigenous soil N and fertilizer N (Cassman et al., 1996b). This parameter for quantifying fertilizer N use efficiency does not need N-omission plots.

The high rates of N fertilizer input and improper timing of N application in China have led to low recovery efficiency and agronomic N use efficiency. Zhu (1985) reported that recovery efficiency in China was less than 30% for ammonium bicarbonate and 30– 40% for urea. Li (1997) estimated that recovery efficiency for rice in China was around 30–35%. However, Li (2000) observed that the average recovery efficiency of rice in Jiangsu Province was only 20%. This was further confirmed by Wang et al. (2001), who reported that recovery efficiency of farmers' N fertilizer practice was 18% in an on-farm experiment conducted in Zhejiang. Peng et al. (2006) found that recovery efficiency of farmers' N fertilizer practice was 20 to 30% in four provinces in China.

Yoshida (1981) estimated agronomic N use efficiency to be 15–25 kg rough rice per kg applied N in the tropics. Cassman et al. (1996b) reported that agronomic N use efficiency was 15 to 18 kg kg<sup>-1</sup> N in the dry season in the farmers' fields in the Philippines. In China, agronomic N use efficiency was 15– 20 kg kg<sup>-1</sup> N from 1958 to 1963 and declined to only 9.1 kg kg<sup>-1</sup> N between 1981 and 1983 (Lin, 1991). Since then, agronomic N use efficiency has further decreased in China because of the increase in N rate (Peng et al., 2002). Wang et al. (2001) reported that agronomic N use efficiency of farmers' N fertilizer practice was  $6.4 \text{ kg kg}^{-1}$  in Zhejiang. Peng et al. (2006) reported that rice yield increases by only 5 to 10 kg for every kg of N fertilizer input by using farmers' N fertilizer practice in China.

# 2 Principles and Procedures of Site-Specific N Management (SSNM)

Site-specific N management (SSNM) was developed to increase fertilizer N use efficiency of irrigated rice (Dobermann et al., 2002). In SSNM, N application is based on the crop demand for N. Climatic factors (solar radiation and temperature) and indigenous N supply largely affect crop N demand. Indigenous N supply includes N from soil mineralization, irrigation water and crop residues. Soil chemical analysis is not reliable to quantify indigenous N supply for paddy soil (Cassman et al., 1996a). Therefore, the measurement of grain yield in N-omission plots is used to obtain field-specific estimates of the indigenous N supply (Cassman et al., 1996b). During the growing season, leaf N status measured with a chlorophyll meter (SPAD) or leaf color chart is a good indicator of crop N demand (Peng et al., 1996). Both the SPAD and leaf color chart provide a good estimation of leaf N content on a leaf-area basis (Peng et al., 1993; Yang et al., 2003). In-season upward or downward adjustments of predetermined N topdressings at critical growth stages are made based on SPAD or leaf color chart readings at these stages. In this approach, the timing and number of N applications are fixed while the rate of N topdressing varies across seasons and locations.

In current SSNM practice, four steps are needed to estimate total N rate based on indigenous N supply capacity and target yield: (1) set an attainable yield target based on 85% of yield potential, (2) estimate indigenous N supply – yield without N fertilizer, (3) estimate N response – the difference between target yield and yield without N fertilizer, and (4) estimate N rate based on N response and agronomic N use efficiency. A realistic target agronomic N use efficiency is chosen by considering the yield and agronomic N use efficiency of the location in previous seasons (IRRI, 2006). The total N is distributed at one day before transplanting, midtillering, panicle initiation and heading with approximate proportions of 35, 20, 30 and 15%, respectively (Peng et al., 2006). The pre-determined N rate is 30 kg ha<sup>-1</sup> at midtillering and 40 kg ha<sup>-1</sup> at panicle initiation. The actual rates of N topdressing at midtillering and panicle initiation are adjusted by  $\pm 10$  kg N ha<sup>-1</sup> according to leaf N status measured with a SPAD or leaf color chart. When leaf N is below the low level of the SPAD or leaf color chart thresholds, N rate will be increased by 10 kg ha<sup>-1</sup>; when leaf N is greater than the upper level of the SPAD or leaf color chart thresholds, it will be decreased by 10 kg ha<sup>-1</sup>. No adjustment in N rate is needed when leaf N is within the thresholds. At heading, N is applied only if leaf N is below the SPAD or leaf color chart thresholds. We used the same SPAD and leaf color chart thresholds across different growth stages for a given variety (Peng et al., 1996). The SPAD and leaf color chart thresholds vary with varieties, which needs to be determined experimentally. Japonica generally requires higher thresholds than indica. For example, SPAD thresholds of 35-37 were used for indica varieties and they were 2 units higher for japonica varieties (Huang et al., 2008). The SPAD value of 35 corresponded to a leaf color chart score of 3.2, 4.8 and 5.2 for the leaf color charts developed by the International Rice Research Institute (IRRI), Zhejiang Agricultural University, China, and University of California-Davis, USA, respectively (Yang et al., 2003).

The procedures of SSNM can be simplified in two ways. Firstly, if there is no N-omission plot and its grain yield is unknown, one can estimate the N response based on the information of climatic yield potential and soil fertility (i.e. skipping the first two steps in estimating total N rate). Secondly, if farmers do not want to measure leaf N status using the SPAD or leaf color chart, upward or downward adjustments of N topdressings can be made based on visual leaf N status (IRRI, 2006).

Since 1997, SSNM has been evaluated in farmers' fields in eight major irrigated rice domains in Asia (Dobermann et al., 2002), including rice farms in Zhejiang Province, China (Wang et al., 2001, 2004). Across all sites in Asia, average grain yield increased by 11% and average recovery efficiency increased from 31% to 40%, with 20% of all farmers achieving more than 50% recovery efficiency (Dobermann et al., 2002).

# 3 IRRI-China Collaboration on SSNM Research

IRRI scientists in collaboration with partners in China have started research work on improving rice N fertilization through SSNM in China since 1997 (Wang et al., 2001). Field experiments were conducted initially in Zhejiang province in 1997 and gradually expanded to Guangdong, Hunan and Jiangsu provinces in 2001, to Hubei province in 2003, and to Heilongjiang province in 2005 (Peng et al., 2006). These six provinces occupy about 45% of the rice planting area in China. Collaborating organizations in China are Zhejiang University, Guangdong Academy of Agricultural Sciences, Hunan Agricultural University, Yangzhou University, Huazhong Agricultural University, Northeast Agricultural University, China Agricultural University and the Chinese Academy of Sciences.

From 1997 to 2000, on-farm demonstration of SSNM was conducted in Zhejiang province in farmers' fields by comparing SSNM with farmers' N fertilizer practice (Wang et al., 2001, 2004). From 2001 to 2003, SSNM was compared with other N treatments including real-time N management, farmers' N fertilizer practice, modified farmers' N fertilizer practice, and three fixed-N split treatments in the replicated on-farm trials in Zhejiang, Guangdong, Hunan and Jiangsu provinces (Peng et al., 2006). In real-time N management, N was applied only when the leaf N content was below a critical level. The farmers' N fertilizer practice was modified by reducing the total N rate by 30% and this reduction was restricted to within 10 days after transplanting. The three fixed-N split treatments had total N rates of 60, 120 and 180 kg ha<sup>-1</sup> with 35% applied at basal, 20% at midtillering, 30% at panicle initiation, and 15% at heading. From 2005 to 2007, SSNM was compared with the real-time N management using four SPAD thresholds in the replicated onfarm trials in Guangdong, Hunan, Hubei and Jiangsu provinces. Overall, SSNM performed better than the real-time N management and the modified farmers' N fertilizer practice because the total N rate of SSNM was closer to the optimal level. In on-farm demonstration and farmer participatory research, we focused mainly on SSNM technology for improving fertilizer N use efficiency in irrigated rice in China.

On-farm demonstration of SSNM was done in the six provinces from 2003 to 2007. Only one location was chosen in each province. About 10 farmers from each location participated in the demonstration experiment. Each farmer's field had three treatments: farmers' N fertilizer practice, SSNM and N omission. The field was divided into two equal parts with a levee, and farmers' N fertilizer practice and SSNM treatments were randomly assigned to each half of the field. The N-omission plot ( $5 \text{ m} \times 5 \text{ m}$ ) was installed in the half of the field with SSNM treatment. In the farmers' N fertilizer practice, farmers apply N fertilizer (form, rate and timing) based on their own decision. Other crop management practices including P and K application were identical among the three treatments.

We used farmer participatory research for on-farm testing of N fertilization by standard and farmermodified SSNM for irrigated rice in Guangdong, Hunan, Hubei and Jiangsu provinces from 2003 to 2005 (Hu et al., 2007). Twelve to 15 farmers were randomly selected in each study village in each year for a dialogue with the research team and for a rapid rural technology assessment. Based on the information obtained from the rapid rural technology assessment, modified SSNM schemes were developed through dialogue between a research team and farmers at a workshop in each village. Modification mainly involved decreasing the number of N fertilizer topdressings and increasing the rate of basal N application. Based on the farmers' willingness, 144 farmers were selected to conduct an experiment to compare SSNM or modified SSNM with farmers' N fertilizer practice.

Data from on-farm field trials, on-farm demonstration and farmer participatory research in the six provinces in China from 2001 to 2007 were pooled to compare SSNM with farmers' N fertilizer practice and other N treatments. The same dataset was used to determine the relationships among yield with N fertilizer, zero-N yield, N response and agronomic N use efficiency. There were 544 observations for these analyses. Data from the on-farm demonstration of SSNM in Zhejiang province from 1997 to 2000 were published elsewhere (Wang et al., 2001, 2004) and not included in the analyses of this paper.

## 4 Key Research Findings

There were wide ranges in both grain yield and total N rate across all experiments (Fig. 1). Grain yield varied from 4 to 10 t  $ha^{-1}$  with an average of 7.28 t  $ha^{-1}$ ,





**Fig. 1** Grain yield and N rate of site-specific N management (solid symbols) and other N treatments (open symbols). Data are from on-farm field trials, on-farm demonstration and farmer participatory research conducted in six provinces in China from 2001 to 2007

while total N rate ranged from 50 to 400 kg ha<sup>-1</sup> with a mean of 157 kg ha<sup>-1</sup>. SSNM was able to narrow the range of total N rate to 80–200 kg ha<sup>-1</sup> with a mean of 120 kg ha<sup>-1</sup>. There was no correlation between grain yield and total N rate. This poor relationship was caused by many factors including location, season, variety, pest damage, other crop management practices, etc., which affected grain yield aside from total N rate.

We observed a relatively high indigenous N supply capacity in irrigated rice fields in China compared with other major rice-growing countries. Crop N uptake at maturity when N fertilizer was not applied was in the range of 90–100 kg ha<sup>-1</sup>, compared with 60–70 kg ha<sup>-1</sup> under tropical conditions (Dobermann et al., 2003). Grain yield of zero-N control ranged from 3 to 8 t ha<sup>-1</sup> and averaged at 5.77 t ha<sup>-1</sup> (Fig. 2). There was a positive relationship between grain yields with and without N fertilizer.

Averaged across 107 farmers who conducted onfarm demonstration, SSNM produced 5% higher grain yield than farmers' N fertilizer practice (Table 1). The yield increase was associated with the reduction





**Fig. 2** Relationship between grain yield with and without N fertilizer. Data are from on-farm field trials, on-farm demonstration and farmer participatory research conducted in six provinces in China from 2001 to 2007. y = 2.16 + 0.886x (r = 0.70; n = 544)

in insect and disease damage and improved lodging resistance of rice crop under the optimal N inputs (Zhong et al., 2006a, b). This small yield increase was achieved when total N rate was reduced from 195 to 133 kg ha<sup>-1</sup>, a 32% saving of N fertilizer over farmers' N fertilizer practice. The N response of SSNM was 28% higher than that of farmers' N fertilizer practice. Consequently, SSNM almost doubled farmers' N fertilizer practice in agronomic N use efficiency, and had 55% higher partial factor productivity of applied N than farmers' N fertilizer practice. Similar results were obtained when SSNM was compared with other N treatments such as real-time N management, farmers' N fertilizer practice, modified farmers' N fertilizer practice, and fixed-N split treatments in 25 replicated on-farm trials in five provinces between 2001 and 2007 (Table 2). Yield increase was insignificant but SSNM saved 25% N fertilizer over other N treatments. SSNM increased N response by 15%, agronomic N use efficiency by 55%, and partial factor productivity of applied N by 38% compared with other N treatments. In farmer participatory research, the rate and distribution of N fertilizer during the growing season of

<b>Table I</b> Grain yield, total N rate, yield response to N appli-	demor
cation, agronomic N use efficiency, and partial factor produc-	China
tivity of applied N (PFP) of farmers' fertilizer practice and	contro
site-specific N management (SSNM). Data were from on-farm	
Parameters Farmers'	practice

. . . . . .

demonstrations conducted by 107 farmers from six provinces in China between 2003 and 2007. Average grain yield of zero-N control was  $5.69 \text{ t ha}^{-1}$  across the 107 farmers

Parameters	Farmers' practice	SSNM	Difference (%)
Grain yield (t ha <sup>-1</sup> )	7.08 b	7.47 a	5
N rate (kg ha <sup><math>-1</math></sup> )	195 a	133 b	38
N response (t $ha^{-1}$ )	1.39 b	1.78 a	25
Agronomic N use efficiency (kg kg <sup><math>-1</math></sup> )	7.1 b	13.4 a	61
$PFP (kg kg^{-1})$	36.3 b	56.2 a	43

Within a row, means followed by different letters are significantly different at the 0.05 probability level according to the least significant difference (LSD) test.

**Table 2** Grain yield, total N rate, yield response to N appli-cation, agronomic N use efficiency, and partial factor produc-tivity of applied N (PFP) of fixed N-rate treatments and site-specific N management (SSNM). Data were from 25 on-farm

field experiments conducted in five provinces in China between 2001 and 2007. Average grain yield of zero-N control was 5.87 t  $ha^{-1}$  across the 25 experiments

Parameters	Other N treatments	SSNM	Difference (%)
Grain yield (t $ha^{-1}$ )	7.45 a	7.69 a	3
N rate (kg ha <sup><math>-1</math></sup> )	161 a	120 b	29
N response (t ha $^{-1}$ )	1.58 a	1.82 a	14
Agronomic N use efficiency (kg kg <sup><math>-1</math></sup> )	9.8 b	15.2 a	43
$PFP (kg kg^{-1})$	46.3 b	64.1 a	32

Within a row, means followed by different letters are significantly different at the 0.05 probability level according to the least significant difference (LSD) test.

modified SSNM were in between those of SSNM and farmers' N fertilizer practice (Hu et al., 2007). Both SSNM and modified SSNM, compared with farmers' N fertilizer practice, maintained rice yields with significantly less N fertilizer and no significant increase in total labor input. The reduction in N fertilizer input averaged 48 kg ha<sup>-1</sup> for SSNM and 23 kg ha<sup>-1</sup> for modified SSNM.

Large variation in N response was observed across 544 observations (Fig. 3). The 1.5 t  $ha^{-1}$  average N response across all experiments is very low compared with the average N response of 3.0 t  $ha^{-1}$  observed under tropical conditions (Dobermann et al., 2003). Negative N response occurred when the application of N fertilizer actually reduced grain yield. Yield reduction was often observed under excessive N input due to greater pest damage and lodging. There was no correlation between N response and total N rate. High total N rates did not result in high N response. The N response was correlated positively with grain yield when N was applied (Fig. 4). There was a negative but weak correlation between N response and the grain yield of zero-N control.



**Fig. 3** Relationship between yield response to N application (N response) and N rate. Data are from on-farm field trials, on-farm demonstration and farmer participatory research conducted in six provinces in China from 2001 to 2007

b



**Fig. 4** Relationship between yield response to N application (N response) and grain yield with N fertilizer (a) and grain yield without N fertilizer (b). Data are from on-farm field trials,



**Fig. 5** Relationship between agronomic N use efficiency and N rate. Data are from on-farm field trials, on-farm demonstration and farmer participatory research conducted in six provinces in China from 2001 to 2007

There was a negative relationship between agronomic N use efficiency and total N rate (Fig. 5). Large variation in agronomic N use efficiency was observed



on-farm demonstration and farmer participatory research conducted in six provinces in China from 2001 to 2007

when total N rates were between 80 and 120 kg ha<sup>-1</sup>. When total N rates were greater than 300 kg ha<sup>-1</sup>, agronomic N use efficiency was below 10 kg kg<sup>-1</sup>. Negative agronomic N use efficiency was observed when grain yield was reduced by the application of N fertilizer. Like N response, agronomic N use efficiency was correlated positively with the grain yield of N-applied treatments but negatively with the grain yield of zero-N control (Fig. 6).

Grain yield and total N rate were generally higher in the North than in the South. Grain yield of zero-N control suggests that indigenous N supply was the lowest in Guangdong compared with the other five provinces. The N response and agronomic N use efficiency were the highest in Heilongjiang compared with the other five provinces. Therefore, there is a great potential to improve N management through SSNM in the provinces where total N rate is high and agronomic N use efficiency is low.

The main reason for poor fertilizer N use efficiency of rice crop in China is that most rice farmers apply excess N fertilizer, especially at the early vegetative stage (Peng et al., 2006). High indigenous N supply capacity and low N response were not considered by rice researchers and extension technicians when N fertilizer rate was determined for recommendation to rice farmers in China. This explains why SSNM usually have



**b** Agronomic N use efficiency (kg/kg)



Fig. 6 Relationship between agronomic N use efficiency and grain yield with N fertilizer (a) and grain yield without N fertilizer (b). Data are from on-farm field trials, on-farm

a lower total N rate than farmers' N fertilizer practice in China. A twelve-season long-term field experiment demonstrated that a decrease in total N rate may reduce indigenous N supply capacity, but will not cause yield reduction in subsequent rice crops as long as optimal N management is practiced (data not shown).

# 5 Remarks on Site-Specific N Management

Improved N management such as SSNM increases both grain yield and fertilizer N use efficiency compared with farmers' N fertilizer practice in China. This is achieved by reducing total N rate and by reducing N rate during the early vegetative stage. The total N rate is reduced because of small N response and high target agronomic N use efficiency. The reduction in N rate during the early vegetative stage is considered because irrigated rice soils in China have a high indigenous N supply capacity, which provides sufficient N for the early vegetative growth. In implementing the SSNM procedure, we have two steps to determine the right N rate for rice crop. The first step is done before crop establishment. The total N rate is estimated based on N response and target agronomic N use efficiency. The percentage of the first N application is demonstration and farmer participatory research conducted in six provinces in China from 2001 to 2007

decided based on previous experimental results and indigenous N supply capacity. The second step is the upward or downward adjustments of topdressing N rate by 10 kg N ha<sup>-1</sup> at critical growth stages based on leaf N status.

Several limitations are associated with SSNM. First of all, grain yield of zero-N control varies with cropping history, variety, climatic conditions and crop management practices. As a consequence, N response is not very stable for a given location across seasons. Secondly, agronomic N use efficiency is affected by many factors such as N response and total N rate. Therefore, total N rate determined by N response and agronomic N use efficiency provides only an approximate range. Thirdly, varietal differences in tillering capacity and early growth vigor are not considered when the percentage of the first N application is decided. Finally, the in-season adjustment of N rate by  $\pm 10$  kg ha<sup>-1</sup> may be inadequate because more adjustment could be needed when yield level and the magnitude of N response are high. Because of these limitations, only 3-5% increase in grain yield can be achieved by SSNM in China. Furthermore, SSNM improved agronomic N use efficiency only up to an average of 15 kg kg<sup>-1</sup> in China. Future research should focus on accurate determination of N response and agronomic N use efficiency using weather data such as solar radiation and air temperature, and information on variety and soil

characteristics. Experiments are needed to test if inseason adjustment of N rate should be linked with the difference between actual SPAD readings and SPAD threshold values.

Several problems exist in the adoption of SSNM technology. First of all, information on the yield of N-omission plots and target yield are not available to many farmers. Secondly, N response is variable and largely affected by variety, crop management practices, season and location. Thirdly, some farmers are reluctant to invest time in monitoring leaf N status using the leaf color chart. Many farmers have difficulty in determining leaf color chart readings accurately. Fourthly, frequent changes in varieties require adjustments in leaf color chart threshold values. Finally, the procedures for implementing SSNM could be complicated for many farmers. These problems may be overcome by developing remote sensing technology such as a canopy reflectance sensor or satellite remote sensing for determining the timing and rate of fertilizer N topdressing during the rice-growing period (Xue and Yang 2008).

Despite the above limitations, a decade of collaborative research work on SSNM between IRRI and Chinese scientists has changed the perception of researchers and farmers about N fertilizer management in irrigated rice in China. The following concepts are accepted now by many rice researchers in China: (1) target yield and indigenous N supply capacity should be considered when total N rate is determined; (2) determination of N application rate at basal and during the early vegetative stage should also be based on indigenous N supply; (3) the rate of N topdressing depends on leaf N status and crop N demand, and (4) regional blanket recommendation for N management will not work well. Nowadays, many researchers have developed local N management practices based on SSNM principles and by integration with other management practices. Many farmers who have been exposed to SSNM understand that it is not true that "the higher the N fertilizer input, the higher the grain yield" and that "the greener the leaves, the better the rice crop". They know that it is necessary to reduce N input at basal and the early vegetative stage and increase N input at late stages. The dynamic change in leaf N status has been embedded in some farmers' minds. They can judge visually when and how much N should be applied without using the leaf color chart.

## 6 Conclusions

After one decade of research on SSNM in China and other Asian rice-growing countries, we believe SSNM is a matured technology for improving both fertilizer N use efficiency and grain yield of irrigated rice crop. Our study suggests that there is potential for large-scale dissemination of SSNM technology in China. Accurate estimation of N response and selection of the right agronomic N use efficiency are needed for a better performance of SSNM. In-season adjustment in N rate should also consider the differences in location, season and variety. SSNM principles are useful for researchers to develop N management practices for their local conditions. Our challenges are to further simplify the procedure of SSNM and to convince policy-makers of the effectiveness of this technology in order to facilitate a wider adoption of SSNM among rice farmers in China. Future research is needed to develop SSNM based on remote sensing technology so that SSNM can be practiced on a large scale.

## References

- Cassman K.G., Dobermann A., Sta Cruz P.C., Gines G.C., Samson M.I., Descalsota J.P., Alcantara J.M., Dizon M.A., Olk D.C. (1996a) Soil organic matter and the indigenous nitrogen supply of intensive irrigated rice systems in the tropics, Plant Soil 182, 267–278.
- Cassman K.G., Dobermann A., Walters D.T., Yang H.S. (2003) Meeting cereal demand while protecting natural resources and improving environmental quality, Ann. Rev. Environ. Resour. 28, 315–358.
- Cassman K.G., Gines G.C., Dizon M.A., Samson M.I., Alcantara J.M. (1996b) Nitrogen-use efficiency in tropical lowland rice systems: contributions from indigenous and applied nitrogen, Field Crops Res. 47, 1–12.
- Cu R.M., Mew T.W., Cassman K.G., Teng P.S. (1996) Effect of sheath blight on yield in tropical, intensive rice production system, Plant Disease 80, 1103–1108.
- Dobermann A., Witt C., Abdulrachman S., Gines H.C., Nagarajan R., Son T.T., Tan P.S., Wang G.H., Chien N.V., Thoa V.T.K., Phung C.V., Stalin P., Muthukrishnan P., Ravi V., Babu M., Simbahan G.C., Adviento M.A.A. (2003) Soil fertility and indigenous nutrient supply in irrigated rice domains of Asia, Agron. J. 95, 913–923.
- Dobermann A., Witt C., Dawe D., Gines H.C., Nagarajan R., Satawathananont S., Son T.T., Tan P.S., Wang G.H., Chien N.V., Thoa V.T.K., Phung C.V., Stalin P., Muthukrishnan P., Ravi V., Babu M., Chatuporn S., Kongchum M., Sun Q., Fu R., Simbahan G.C., Adviento M.A.A. (2002) Site-specific nutrient management for intensive rice cropping systems in Asia, Field Crops Res. 74, 37–66.

- FAOSTAT (2008) FAO Statistical databases, http://www.fao.org, Food and Agriculture Organization (FAO) of the United Nations, Rome.
- Heffer P. (2008) Assessment of fertilizer use by crop at the global level, International Fertilizer Industry Association (IFA), Paris, France, pp. 1–5.
- Hu R., Cao J., Huang J., Peng S., Huang J., Zhong X., Zou Y., Yang J., Buresh R.J. (2007) Farmer participatory testing of standard and modified site-specific nitrogen management for irrigated rice in China, Agric. Syst. 94, 331–340.
- Huang J., He F., Cui K., Buresh R.J., Xu B., Gong W., Peng S. (2008) Determination of optimal nitrogen rate for rice varieties using a chlorophyll meter, Field Crops Res. 105, 70–80.
- IFA (2002) Fertilizer use by crop, 5th ed. http://www.fertilizer. org/ifa/statistics.asp, International Fertilizer Industry Association (IFA), International Fertilizer Development Center (IFDC), International Potash Institute (IPI), Potash and Phosphate Institute (PPI), and Food and Agriculture Organization (FAO).
- International Rice Research Institute (IRRI) (2006) Site-specific nutrient management, www.irrri.org/irrc/ssnm/, accessed 20 March 2006.
- Lemaire G., Gastal F. (1997) Nitrogen uptake and distribution in plant canopies, in: Lemaire G. (Ed.), Diagnosis of the Nitrogen Status in Crops, Springer-Verlag, Berlin, pp. 3–43.
- Li Q. (1997) Fertilizer issues in the sustainable development of China agriculture, Jiangxi Science and Technology Press, China.
- Li R. (2000) Efficiency and regulation of fertilizer nitrogen in high-yield farmland: A case study on rice and wheat double maturing system agriculture area of Tai lake for deducing to Jiangsu Province, Ph.D. Dissertation, China Agricultural University, Beijing, China.
- Lin B. (1991) Make the most efficient use of fertilizers in increasing crop production, in: Soil Science in China: Present and Future, Soil Science Society of China, Jiangsu Science and Technology Press, China, pp. 29–36.
- Novoa R., Loomis R.S. (1981) Nitrogen and plant production, Plant Soil 58, 177–204.
- Peng S., Buresh R.J., Huang J., Yang J., Zou Y., Zhong X., Wang G., Zhang F. (2006) Strategies for overcoming low agronomic nitrogen use efficiency in irrigated rice systems in China, Field Crops Res. 96, 37–47.

- Peng S., Garcia F.V., Laza R.C., Cassman K.G. (1993) Adjustment for specific leaf weight improves chlorophyll meter's estimate of rice leaf nitrogen concentration, Agron. J. 85, 987–990.
- Peng S., Garcia F.V., Laza R.C., Sanico A.L., Visperas R.M., Cassman K.G. (1996) Increased N-use efficiency using a chlorophyll meter on high-yielding irrigated rice, Field Crops Res. 47, 243–252.
- Peng S., Huang J., Zhong X., Yang J., Wang G., Zou Y., Zhang F., Zhu Q., Buresh R., Witt C. (2002) Challenge and opportunity in improving fertilizer-nitrogen use efficiency of irrigated rice in China, Agric. Sci. China 1, 776–785.
- Pham Q.D., Abe A., Hirano M., Sagawa S., Kuroda E. (2004) Analysis of lodging-resistant characteristics of different rice genotypes grown under the standard and nitrogen-free basal dressing accompanied with sparse planting density practices, Plant Prod. Sci. 7, 243–251.
- Wang G.H., Dobermann A., Witt C., Sun Q.Z., Fu R.X. (2001) Performance of site-specific nutrient management for irrigated rice in southeast China, Agron. J. 93, 869–878.
- Wang G.H., Sun Q., Fu R., Huang X.H., Ding X.H., Wu J., He Y.F., Dobermann A., Witt C. (2004) Site-specific nutrient management in intensive irrigated rice systems of Zhejiang province, China, in: Dobermann A. et al. (Eds.), Increasing Productivity of Intensive Rice Systems through Site-specific Nutrient Management. Science Publishers, Inc., Enfield, NH (USA), and International Rice Research Institute, Los Baños, Philippines, pp. 243–263.
- Xue L., Yang L. (2008) Recommendations for nitrogen fertilizer topdressing rates in rice using canopy reflectance spectra, Biosyst. Eng. 100, 524–534.
- Yang W.-H., Peng S., Huang J., Sanico A.L., Buresh A.L., Witt C. (2003) Using leaf color charts to estimate leaf nitrogen status of rice, Agron. J. 95, 212–217.
- Yoshida S. (1972) Physiological aspects of grain yield, Annu. Rev. Plant Physiol. 23, 437–464.
- Yoshida S. (1981) Fundamentals of rice crop science. International Rice Research Institute, Los Baños, Philippines, 269 p.
- Zhong X., Peng S., Buresh R.J., Huang N., Zheng H. (2006a) Some canopy indices influencing sheath blight development in hybrid rice, Chinese J. Rice Sci. 20, 535–542.
- Zhong X., Zheng H., Huang N., Ou J. (2006b) Performance of site-specific nutrient management in double-season rice in South China, China Rice 3, 30–32.
- Zhu Z. (1985) Research progresses on the fate of soil N supply and applied fertilizer N in China, Soil 17, 2–9.

# Solid–Liquid Separation of Animal Slurry in Theory and Practice

Maibritt Hjorth, K.V. Christensen, M.L. Christensen, and Sven G. Sommer

# Contents

1	Intro	duction	954
2	Slurr	y Production	955
3	Nitro	gen and Phosphorus Transformation during	
	Stora	ge	956
	3.1	Nitrogen	956
	3.2	Phosphorus	956
4	Physic	cal and Chemical Characterization of Slurry .	957
	4.1	Particle Size	957
	4.2	Organic Components	958
	4.3	Inorganic Components	958
	4.4	pH Buffer Systems	959
	4.5	Electrochemical Properties	960
	4.6	Physical Properties	960
5	Solid-	-Liquid Separation	961
	5.1	Sedimentation	962
	5.2	Centrifugation	964
	5.3	Drainage	966
	5.4	Pressurized Filtration	967
	5.5	Concluding Remarks about Slurry Separators	969
6	Pre-ti	reatment: Chemical Additions	969
	6.1	Precipitation, Coagulation and Flocculation	969
	6.2	Struvite Crystallization	977
7	Post-t	reatment: Separation Techniques	979
	7.1	Evaporation	979
	7.2	Membranes	979
8	Conc	usions	981
Re	ference	es	982

e-mail: sgs@kbm.sdu.dk

Abstract Animal slurry contains plant nutrients that are essential for crop production. However, intensive livestock production may lead to a surplus of plant nutrients on farms and, as a consequence, discharge or emission to the environment. In order to ensure that the slurry applied to fields matches the nutrient requirements of the crops, techniques have been developed to reduce the nutrient content of slurry by means of separation. This review discusses the separation technologies used for animal slurry treatment and the physical and chemical processes involved in separation. These processes need to be understood before efficient, reliable and cheap separation technologies that take into account the actual properties of slurry and the likely end-use of the separation products can be developed. A simple separation efficiency expression can be used to assess the efficiency of slurry separation. It is indeed important to measure the amount and composition of the slurry before treatment, the dry-matter-rich fraction and the liquid fraction. The separation efficiency of mechanical separators for the removal of dry matter and phosphorus (P) is ranked as follows: centrifugation > sedimentation > non-pressurized filtration > pressurized filtration. In general, the separation of total N and  $NH_4^+$  follows the same pattern, but the separation efficiency is lower than for dry matter and P. Treatment with a flocculant before separation improves separation efficiency significantly. Of the polyacrylamide polymers tested, high-molecular-weight, linear cationic polymers with a medium charge density (20-40 mol%) were found to be the most efficient flocculants. The best mechanical separation techniques for flocculated slurry are screens or filter belts. The separation efficiency of polyacrylamide-treated slurry can be improved by

S.G. Sommer (🖂)

Department of Biosystems Engineering, Faculty of Agricultural Sciences, Aarhus University, Blichers Allé 20, 8830 Tjele, Denmark

<sup>E. Lichtfouse et al. (eds.),</sup> *Sustainable Agriculture Volume 2*, DOI 10.1007/978-94-007-0394-0\_43,
© Springer Science+Business Media B.V. - EDP Sciences 2011. Reprinted with permission of EDP Sciences from Hjorth et al., Agron. Sustain. Dev. 30 (2010) 153–180. DOI: 10.1051/agro:2009010

adding a multivalent ion to coagulate particles and for precipitation of phosphorus. Aluminium sulfate  $(Al_2(SO_4)_3)$  or ferric chloride (FeCl<sub>3</sub>) seem to be very efficient for improving the mechanical separators. Alternatively, the mineral struvite  $(MgNH_4PO_4)$ may be formed by changing the slurry characteristics, such as by the addition of magnesium (Mg) or by increasing the pH to 9. The struvite crystals are removed during solid-liquid separation. The products of the solid-liquid separation may be further treated by evaporation, membrane filtration or ammonia stripping in order to obtain the desired end-products; however, low-maintenance and/or cost-efficient operation of these post-treatments has not yet been demonstrated. The separation should be developed as a whole-system approach, paying attention to parameters such as the value of end-products, environmental consequences and economy.

**Keywords** Manure • Phosphorus • Nitrogen • Sedimentation • Centrifugation • Filtration • Coagulation • Flocculation • Struvite

# **1** Introduction

Livestock production increased towards the end of the 20th century and, due to increasing demand in developing countries, production is expected to increase further in Asia and South America (FAOSTAT, 2006). Globally, the intensification of agricultural systems increases the environmental impact of food production, as high concentrations of livestock increase the risk that organic waste nutrients are used for crop production in excess of crop requirements.

This section briefly discusses current problems related to discharge and the unsustainable use of slurry, with a focus on how to optimize separation as a sustainable and cost-effective treatment option for animal slurry.

Livestock slurry contains phosphorus (P) and potassium (K), which have fertilizer values equivalent to those of mineral fertilizers, while its nitrogen (N) content has a lower and more variable fertilizer value than that of commercial fertilizers. If applied at a rate higher than plant uptake, there is a great risk of nutrient leaching and runoff that will pollute surface- and groundwater (Burton and Turner, 2003). This leads to an increased need for water purification to provide safe drinking water supplies. Slurry is also harmful to the environment when discharged directly into receiving waters (Rattanarajcharkul et al., 2001; Steinfeld et al., 2006; Petersen et al., 2007; Vu et al., 2007). Negative effects from heavy applications of slurry may include salinization in semi-arid regions, toxic concentrations of heavy metals, and decreased soil aeration (Bernal et al., 1992, 1993). Larger production units may result in higher local emissions of odor and ammonia gas from housing and stores. Intensive livestock production may also lead to higher energy use for the transport of livestock wastes to be recycled in crop production elsewhere (Sørensen et al., 2003; Sørensen and Møller, 2006), and the risk of spreading disease among livestock will increase.

Until recently, European farmers have tried to minimize the environmental problems caused by slurry management through recycling on farm. It has been recognized that livestock slurry represents a valuable resource that, if used appropriately with minimal loss, can replace significant amounts of mineral fertilizers (Bouwman and Booij, 1998; Le, 1998). However, an increasing number of large, intensive livestock production units with insufficient area for the sustainable recycling of slurry nutrients have emerged in recent years. This specialization in livestock production tends to weaken the link between livestock and plant production and increases the farm's impact on the environment. To facilitate the recycling of slurry produced on these farms, there is a need to transport the slurry to farms specializing in crop cultivation. The cost of transporting slurry may be reduced, and its fertilizer value increased, by separating the slurry into (1) a liquid fraction intended for on-farm use, and (2) a dry-matter- and nutrient-rich fraction that can be exported to farms with few or no animals (Møller et al., 2000; Sørensen et al., 2003). Separation may also contribute to a reduction in odor emission (Zhang and Westerman, 1997) and assist in producing energy-rich biomass that can be used for incineration or biogas production (Møller et al., 2007a; Hjorth et al., 2009).

In Asia, slurry from pig farms is often separated in-house on small- to medium-sized farms by manual scraping, while on large farms the solid fraction is separated from the liquid fraction by filter separators or decanter centrifuges (Ong, 2002; Vu et al., 2007). The liquid fraction is transferred to fish ponds, where it fertilizes aquatic plants that are eaten by herbivorous fish (Vu et al., 2007). The solid slurry produced by pigs and cattle that is scraped off the floor is composted, and an organic fertilizer is produced that is sold to vegetable and fruit producers as a highly valued product for soil amelioration.

In Europe and North America, the solid fraction is often not considered to have much value and slurry processing is therefore considered as an additional cost of livestock production (Petersen et al., 2007). Thus, the challenge is to reduce the cost and improve the efficiency of the processes while adding value to the separation products - especially the solid fraction, which has a high energy and P content. In this context, it is important to understand that slurry management consists of several interrelated operations carried out from the time the slurry is removed from the animal house until it is used for bioenergy production and to fertilize crops. Therefore, a whole-system approach should be considered when developing the technology for optimizing the recycling of plant nutrients in livestock production and optimizing green energy production using livestock slurry (Petersen et al., 2007).

It should be borne in mind that the separation of animal slurry may create new problems; for example, a change in the ratio of plant nutrients to heavy metals in the biomass. A recent study showed that solid–liquid separation with flocculation as a pre-treatment transferred zinc (Zn), copper (Cu) and cadmium (Cd) to the solid fraction (Møller et al., 2007b). The additives used when optimizing the separation of slurry (e.g. polymers and Al) may also pose an environmental problem (Nahm, 2005; Schechter et al., 1995).

Overall, slurry separation and recycling of organic matter and plant nutrients can mitigate the environmental hazards described above, since it will contribute to the optimization of crop fertilization, thereby reducing the need for external supplies of N, P, K and micronutrients.

An effective separation system can only be designed if the following three points are taken into account:

- 1. the chemical and physical properties of the slurry;
- 2. the desired end-products;
- the potential separation techniques, including preand post-treatments and combinations of different techniques.

Knowledge of these variables will provide the information needed to produce separation products that can be used efficiently in crop production. Desirable separation products would, for instance, comprise a liquid fraction containing N, P and K at a ratio equivalent to the plant demand, and a solid fraction with high P, a relatively high N content (including organic N), and a high content of DM, which could be used as a carbon source for energy production (Hjorth et al., 2009).

A thorough understanding of the physical and chemical processes involved in the separation of slurry is required in order to develop reliable, efficient and cheap separation technologies, taking into account the actual slurry properties and the end-use of the separation products. This review aims to present a characterization of animal slurry relevant for slurry separation, and to describe concepts of separation optimized to treat slurry with the specified characteristics. The review includes the presentation of separation techniques, including the use of additives as pre-treatments.

# 2 Slurry Production

The management of slurry will affect its physical and chemical characteristics. Animal housing may be designed so that both solid and liquid slurry are produced. Alternatively, some animal houses are designed that have slatted floors with cellars or channels beneath. Below the slats, a slurry consisting of a mixture of feces and urine, strewing material, and spilt feed and water is collected.

The range of animal housing design and methods of slurry collection, storage and handling reflect the large differences in climate and production objectives throughout the world. Slurry management systems have been developed in order to reduce the manual labor required. The animal slurry in these systems is mainly stored and transported from animal houses via deep pit, pull plug, pit recharge and flushing systems (Menzi, 2002; Arogo et al., 2003). There is little or no seasonal variation in untreated slurry composition (Petersen and Kjellerup, 1990; Conn et al., 2007), which reflects the fact that the diets given to the animals and the slurry management normally do not vary significantly throughout the year. With respect to the separation of animal slurry, the slurry removal system and method of storage is important because the physical and chemical characteristics of the slurry

will change due to microbial transformation processes during storage (Hindrichsen et al., 2006). Slurry characteristics may also be changed by introducing additives or changes in feed, as these changes will be reflected in slurry composition (Otto et al., 2003). The composition of the slurry also varies significantly with depth in the slurry store due to sedimentation and surface crust formation (Burton and Turner, 2003).

Separation of urine and feces may be carried out in the animal house. In cattle houses with tethered cattle, the manure may be separated into solid and liquid fractions by collecting the solids from the floor behind the animals while the liquid is drained through gutters. In these houses, the collected solid manure mainly contains feces and straw and the liquid slurry contains a mixture of water, urine and soluble fecal components (Sommer and Hutchings, 2001; Sommer et al., 2007). In modern pig houses, the animals move freely on slatted floors, and therefore in-house separation of the slurry needs to be carried out below the slatted floor (Kroodsma, 1986; Lachance et al., 2004). However, farmers are still reluctant to install this technology because of problems related to equipment maintenance; instead, the separation of mixed urine and feces, i.e. slurry, will be the focus of this review.

# 3 Nitrogen and Phosphorus Transformation during Storage

The distribution of nitrogen (N) and phosphorus (P) between the solid and liquid slurry fractions is affected by the microbial transformation of organic matter that takes place during storage before slurry separation. As shown in Sect. 4, the N and P are distributed between a solid phase (particles) and a liquid phase, and this distribution is affected by excretion and storage.

## 3.1 Nitrogen

In livestock farming, usually not more than 5–45% of the nitrogen (N) in plant protein is transformed into animal protein, depending on the type of animal and livestock management (Oenema et al., 2001). The remaining 55–95% is excreted via urine and dung as organically bound N. Following deposition on the floor of the animal house or in pastures, a major fraction of the organic N is rapidly hydrolyzed into ammonium  $(NH_4^+)$  by the enzyme urinase, which is present in the feces (Sommer et al., 2006).

Immobilization of inorganic N into organically bound N is a microbial process, which depends on the C:N ratio in the degradable organic compounds. Slurry mixtures have C:N ratios in a range between 4 for pig slurries and 10 for cattle slurries (Chadwick et al., 2000). In general, there is no immobilization of N in slurry mixtures, because the C:N ratio of the easily degradable compounds is low (<15) (Kirchmann and Witter, 1989; Thomsen, 2000). During in-house storage, most of the digestible compounds containing N are transformed, and about 10% of the organic N is mineralized (Zhang and Day, 1996; Sørensen, 1998). During outdoor storage of slurry, little N is mineralized and it is assumed that about 5% of the organic N is transformed into inorganic N during the course of 6-9 months' storage (Poulsen et al., 2001).

Approximately 70% of the N in slurry is dissolved and most of the dissolved N is present as  $NH_4^+$ (Christensen et al., 2009). The  $NH_4^+$  in slurry, as well as in the liquid and solid separation fractions, may be lost due to volatilization as  $NH_3$  (pKa = 9.25) or after microbial transformation as nitrous oxide or N<sub>2</sub> (Hansen et al., 2006). Thus, the total loss of gaseous N from animal housing and slurry storage ranges from about 10% of the excreted N in dung and urine from dairy cattle to >30% for pigs and poultry in intensive livestock operations (Oenema et al., 2001).

## 3.2 Phosphorus

Phosphorus is fed to the animals via plants, but also as additives in the form of inorganic phosphate. Cattle are able to utilize P very efficiently and only the plant P that is not mobilized and adsorbed in the digestive system is excreted. Most of this P is excreted in feces and very little in urine (Meyer et al., 2007). Higher amounts of dietary P may lead to higher P excretion in feces (Dou et al., 2003; Chapuis-Lardy et al., 2004).

The digestive system of pigs is not as efficient as that of cattle in absorbing P in feed, and pigs excrete about 50–60% of their P intake in feces and urine. Surplus P intake is excreted in urine. The enzyme phytase is added to pig feed to improve P uptake. Phytase degrades the organic P compound phytate and makes the P more easily available; consequently, the inorganic P supplement can be reduced and P excretion reduced significantly, i.e. by >20% (Pfeiffer, 1995; Han et al., 2001; Sommer et al., 2008). This reduction mostly affects the amount of P in urine (Poulsen, 2000) and is affected by P digestibility, the amount of feed in relation to the requirements of the pig, and the age and category of animal. Production systems will also affect P excretion; for instance, changing the group size of fattening pigs may increase welfare and reduce P excretion. In contrast, group housing of sows and the addition of straw may increase P excretion (Krieter, 2002).

In slurry >80% of the dissolved P is orthophosphate. The orthophosphate fraction may vary during storage, i.e. increase immediately after excretion and reach a maximum, and thereafter it will decrease slightly (Christensen et al., 2009).

# 4 Physical and Chemical Characterization of Slurry

Animal slurry from housing is a mixture of feces and urine, bedding material (e.g. straw, wood shavings, sawdust, sphagnum), spilt feed and drinking water, and water used for washing floors. Factors of importance for optimizing the efficiency of slurry separation include particle size, concentration of organic and inorganic components, pH and buffer systems of the slurry. The physical and electrochemical properties of the particles are also important for separation.

Water use will vary due to differences in animal housing systems and management; therefore, due to the amount of dilution, the composition of the animal slurry will vary. In pig slurry, in particular, the concentration of the components varies due to variations in use of water, for instance, because more water is used in sow than in pig houses (Conn et al., 2007), and because a large amount of water is used in Asian pig production to cool the pigs and to clean solid floors (Taiganides, 1992; Sommer et al., 2005). Thus, in slurry from finishing pig houses, the total N, P and K concentrations may be 2.3, 2.5 and 1.7 times greater, respectively, than in sow slurry (Conn et al., 2007). Also, the average *DM* content in slurry from

finishing pigs is six times greater in European slurry than in Asian slurry (Sommer et al., 2005).

# 4.1 Particle Size

Particle size distribution in the slurry is important for separation; for example, filtration will only retain particles above a certain size, and during sedimentation particles in the range between 1 nm and 1  $\mu$ m (i.e. colloids) are subject to Brownian motion and move by diffusion in the liquid, and therefore settle very slowly or not at all.

The amount of *DM* in the particle size fraction below 0.025 mm is greater in pig slurry than in cattle slurry; 66–70% and 50–55% in pig and cattle slurry, respectively (Møller et al., 2002; Sommer et al., 2008). Feed composition may affect particle size distribution; for instance, the proportion of *DM* in the particle size fraction >1 mm was lower in slurry from cows fed only roughage compared with slurry from cows whose diet was supplemented with concentrates (Nørgaard, 2006; Sommer et al., 2008). The effects of diet and category of animal are also reflected in the variation in total solids found in the <0.125 mm category of fresh feces from cattle, being 50%, 41% and 44% (±1.4) for lactating cows, heifers and calves, respectively (Meyer et al., 2007).

Microbial transformation of the organic pool (see Sect. 3) will change the particle size distribution. The amount of organic matter in slurry decreases due to microbial transformation into CH<sub>4</sub> and CO<sub>2</sub> during anaerobic storage. The *DM* content may decrease by 25% during 5 months of storage at 20 °C (Møller et al., 2002). At low storage temperatures, anaerobic transformation of organic material – and therefore *DM* reduction – will be significantly lower (Christensen et al., 2009), i.e. about 10 times lower at 10 °C than the reduction at 20 °C (Sommer et al., 2006).

Anaerobic digestion reduces the *DM* concentration of animal slurry, and changes the particle size distribution. It has been observed that particles <10  $\mu$ m account for 64% of *DM* in raw slurry, while it increases to 84% of *DM* in anaerobically digested slurry (Massé et al., 2005). In contrast, the concentration of particles <1.6  $\mu$ m is reduced most during anaerobic storage in slurry stores (Møller et al., 2002). In anaerobic digesters, stirring and high

temperatures facilitate microbial degradation of large particles, leaving relatively small particles with slowly degradable material. In a cold unstirred slurry, the slow transformation of particles primarily degrades small particles, transforming only a minor fraction of the large particulate material.

In cattle slurry >80% of N and P is in the particle fraction below 0.125 mm (Meyer et al., 2007), and a very detailed analysis of particle size fractions showed that in slurry more than about 70% of the undissolved N and P was in the particle size fraction 0.45–250  $\mu$ m (Massé et al., 2005). Because approximately 30% and 80% of N and P, respectively, is contained in particles (see Sect. 3), the small particles contain a large percentage of the total P in slurry. The addition of phytase to the diet of pigs or supplementing the diet of dairy cows with concentrates does not affect the distribution of total P in the different particle size fractions (Sommer et al., 2007).

# 4.2 Organic Components

The composition of the organic components may affect separation, as charge and pH will influence flocculation and also the cation absorption on and charge neutralization of the particles being separated from the liquid. Animal slurry contains short-chain volatile fatty acids (VFA) with 1–5 atoms of carbon, organic lipids, proteins, carbohydrates and lignin, which also contains non-carbohydrate organic components such as phenols (Møller et al., 2004; Massé et al., 2005; Christensen et al., 2009). Carbohydrates comprise the largest fraction of the organic material, followed by proteins, lipids, lignin and VFA. The organic components in slurry include compounds with the functional groups carboxylates, hydroxyls, sulfur hydryls and phenols (Massé et al., 2005) which, at the pH interval in slurry, will contribute to a negative charge of both dissolved and particulate organic matter.

Most of the P in animal slurry is in the particle fraction of slurry, and >30% is dissolved in the liquid phase (Fig. 1). One study showed that organic P in solution only constitutes about 5% of the P in slurry (Fordham and Schwertmann, 1977a). The remaining P is mainly associated with the particle fraction, where it may be bound inorganically in crystalline form or is adsorbed onto particles.

Total phosphor, P  $\mu$ g g<sup>-1</sup>(DM)



**Fig. 1** Distribution of phosphorus between the particles, colloids and dissolved fraction of pig slurry (adapted from Christensen et al., 2009)

#### 4.3 Inorganic Components

The separation efficiency of different technologies is greatly affected by the slurry composition; for instance, large minerals will end up in the solid fraction. In addition, electrical conductivity will affect flocculation, and the relative concentration of cations will affect crystallization of compounds such as struvite.

The composition of ions in animal slurry varies significantly among slurry from different animals and is affected by the diets fed to the animals and slurry management (Table 1). Electrical conductivity depends on the concentration and species of ions in solution and varies from 0.008 to 0.026 S cm<sup>-1</sup> (Sommer and Husted, 1995a; Christensen et al., 2009). Cattle slurry contains more sodium, potassium and calcium than pig slurry (i.e. 2–3 times more) due to an intake of roughage having a high content of these cations (Sommer and Husted, 1995a; Massé et al., 2007a). In cattle slurry, the content of these ions varies twofold due to variation in diets (Chapuis-Lardy et al., 2004).

The distribution of P between the solid and liquid phases is affected by pH, with more P being dissolved at a decreased pH. The distribution is almost constant down to pH = 6.5, whereas the concentration of dissolved P increases by a factor of 5 when pH is lowered from 6.5 to 5.5; mainly due to dissolution of struvite (Christensen et al., 2009). In slurry, P may be crystallized as struvite (MgNH<sub>4</sub>PO<sub>4</sub> × 6H<sub>2</sub>O) and different forms of calcium phosphates (Fordham

		Comp	onent cont	ent (g/kg s	lurry)						
Slurry origin		DM	Total-P	Total-N	TAN	Κ	Ca	Mg	Cu	Zn	Reference
Sows	Mean Standard deviation	23 (15)	0.8 (0.2)	3.2 (0.9)	2.0 (0.7)	2.0 (0.3)	0.7 (0.4)	0.2 (0.1)	0.0		Møller et al., 2007a; Conn et al., 2007
Finishing pigs	Mean Standard deviation	67 (26)	2.1 (0.8)	7.5 (2.5)	4.5 (2.1)	3.3 (1.1)	2.1 (0.9)	0.9 0.4	0.0 0.0	0.1 (0.1)	Conn et al., 2007; Ye et al., 2005; Massé et al., 2005; Møller et al., 2004, 2007a; Saeys et al., 2005; Sommer and Husted, 1995a; Sommer et al., 2008
Dairy cows	Mean Standard deviation	82 (24)	1.0 (0.2)	3.7 (1.7)	5.0 (9.3)	4.3 (1.8)	1.6	0.6	0.0		Amon et al., 2006; Clemens et al., 2006;Møller et al., 2007a; Sommer and Husted, 1995a; Sommer et al., 2008

**Table 1** The composition of animal slurries is most variable both between samples of slurries from one site and between slurries from different sites

and Schwertmann, 1977b, c), i.e. calcium dihydrogen phosphate  $Ca(H_2PO_4)_2$ , calcium hydrogen phosphate  $CaHPO_4$  and tricalcium phosphate  $Ca_3$  (PO<sub>4</sub>)<sub>2</sub>. The forms of phosphate will be affected by the pH of the slurry (Eq. (1)).

$$Ca_{3}(PO_{4})_{2} + 2H_{2}SO_{4} + 4H_{2}O \Leftrightarrow$$

$$2CaSO_{4} \cdot 2H_{2}O + Ca(H_{2}PO_{4})_{2} \cdot 2H_{2}O \quad (1)$$

Studies suggest that, in slurry, phosphate can crystallize as struvite (Fordham and Schwertmann, 1977b, c; Bril and Salomons, 1990; Sommer and Husted, 1995a; Gungor and Karthikeyan, 2005). Ammonium is present in large excess for struvite formation, thus crystallization is mainly controlled by magnesium concentration and pH.

An increase in the amount of iron  $(Fe^{3+})$  dissolved in the slurry will reduce pH; for example, if the  $Fe^{3+}$ precipitates as crystalline iron phosphate.

$$\operatorname{Fe}^{3+} + \operatorname{H}_3\operatorname{PO}_4^{n-3} \leftrightarrow \operatorname{FePO}_4(s) + n\mathrm{H}^+$$
 (2)

Precipitation of calcite (CaCO<sub>3</sub>) is mainly controlled by the concentration of calcium (Ca<sup>2+</sup>), because carbonate (CO<sub>3</sub><sup>2-</sup>) is present in large excess in slurry. Slurry pH is reduced significantly by adding Ca<sup>2+</sup>, thereby precipitating CO<sub>3</sub><sup>2-</sup> and producing the crystal calcite (Witter and Kirchmann, 1989; Husted et al., 1991).

$$Ca^{2+} + HCO_3^- + H^+ \rightleftharpoons CaCO_3(s) + 2H^+ \qquad (3)$$

Part of the inorganic P may precipitate as calcium phosphate, or it may absorb to CaCO<sub>3</sub> (Fordham and Schwertmann, 1977a, b).

Most  $K^+$  and  $NH_4^+$  salts are very soluble, and struvite (see Sect. 6.2) is the only crystal that in practice contributes to settling or separation of ammonium (Bril and Salomons, 1990). Thus, most  $K^+$  and  $NH_4^+$  is dissolved in the liquid phase (Massé et al., 2005).

#### 4.4 pH Buffer Systems

When considering which process or technology is optimal for the separation of slurry, the pH needs to be known, since struvite precipitation is greatly affected by pH, as is the precipitation of P and ammonia stripping. It has been shown that the main buffer components in animal slurry controlling [H<sup>+</sup>] are total inorganic carbon (TIC =  $CO_2$ +HCO<sub>3</sub><sup>-</sup> + H<sub>2</sub>CO<sub>3</sub>), total ammoniacal nitrogen (TAN = NH<sub>3</sub>+NH<sub>4</sub><sup>+</sup>) and volatile fatty acids (VFA = C<sub>2</sub>-C<sub>5</sub> acids) (Angelidaki et al., 1993; Sommer and Husted, 1995a, b; Vavilin et al., 1998). In stored slurry, the concentration of TIC may be larger than those of TAN and VFA; so this component is therefore of major importance in controlling slurry pH (Japenga and Harmsen, 1990; Sommer and Husted, 1995b). In addition to inorganic buffers, negatively charged particles in the slurry will contribute to acid groups with pKa values from 5 to 9 (Christensen et al., 2009).

A number of processes will affect the pH of slurry; the emission of CO<sub>2</sub> and NH<sub>3</sub> is particularly important because emission of CO<sub>2</sub> will increase pH, and emission of NH<sub>3</sub> will reduce pH. Oxic degradation of organic material reduces the content of acids in solution and thereby increases pH. In contrast, anoxic processes will contribute to the formation of organic acids (e.g. VFA = C<sub>1</sub>-C<sub>5</sub>) and thereby reduce pH. In an anoxic environment, the content of organic acids may be reduced by methanogenic microorganisms by transformation into CH<sub>4</sub> and CO<sub>2</sub> which, in contrast, will increase pH.

Diet and feeding practices may affect the concentration of ionic species in the slurry and the pH, because the electric charge of the solution has to be neutral. At present, soya beans in the diet supply most of the crude proteins needed by pigs. As soya contains high concentrations of  $K^+$ , this will, when excreted, increase the pH of urine and slurry. Reducing the soya concentration in the diet and supplementing with amino acids will reduce the K<sup>+</sup> concentration and, because of the charge balance, the concentration of H<sup>+</sup> will increase (Sommer and Husted, 1995b). Thus, for pig urine and slurry, and for cattle urine, it has been shown that pH declines when cationic species in the feed are reduced; for pig slurry, a reduction of more than 1 pH unit has been observed within the range of traditional diets with and without the addition of amino acids and the reduction of soya (Canh et al., 1998; Portejoie et al., 2004).

The addition of multivalent ions (e.g. FeCl<sub>3</sub> or  $Al_2(SO_4)_3$ ) or cations precipitating carbonate ions (Ca(OH)<sub>2</sub>) will also affect slurry pH, as indicated in Section 4.3 (Husted et al., 1991; Henze et al., 1997; Estevez Rodríguez et al., 2005).

## 4.5 Electrochemical Properties

Particle charge and ionic strength, which affect the electrical potential around the particles, are of vital importance for the efficient use of flocculants.

Organic particles often have a negative surface charge and the particles will therefore repel each other (Gregory, 1989). It has been shown that, in pig slurry, the particle charge density is  $-0.18 \text{ meg g}^{-1}$  organic solids (Christensen et al., 2009). Furthermore, due to their negative charge, organic particles contribute to the alkalinity of the slurry. The concentration of ions expressed as ionic strength has been shown to be high in most animal slurry studies, i.e. conductivity  $>10 \text{ mS cm}^{-1}$  (Sommer and Husted, 1995b; Christensen et al., 2009). An increasing ionic strength of the liquid will affect processes such as flocculation due to a reduced electrostatic repulsion. Furthermore, at high ionic strength, a linear polymer will change to a more sphere-shaped form, which will lower the efficiency of the polymer (Gregory, 1973, 1989).

The negative charge of organic particles in animal slurry affects the distribution of the ions in solution by adsorbing cations close to the particles (Stern layer), with the remaining counter-ions being distributed in the diffusive layer, which also contains anions. These two layers of ions are referred to as the 'double layer'. It is assumed that the interaction between charged particles is visualized as the overlap of diffusive layers of particles. Therefore, the electrical potential of interest, when considering interactions between two particles, is the potential at the boundary between the Stern and the diffusive layers, which is assumed to be equal to the electrokinetic or zeta potential ( $\zeta$ , mV; Lyklema, 1977). The effect of the particle charge density will be affected by the extent of the diffusive layer, which decreases with increasing ionic strength. The influence of the particle charge density is therefore expected to be low in slurry with a high ionic strength.

# 4.6 Physical Properties

The density and viscosity of slurry has been correlated to the dry matter (DM) content. Landry et al. (2004) give a thorough review in connection with their own work on concentrated slurry. Their conclusion is that, for a *DM* between 0 and 50%, the density can be described by the following equations: Dairy cattle:

$$\rho = 1000 + 14.6 \cdot DM - 2.38 \cdot DM^2 + 0.0367 \cdot DM^3$$
(4)

Pig:

$$\rho = 1000 - 11.2 \cdot DM + 1.19 \cdot DM^2 - 0.0235 \cdot DM^3$$
(5)

where  $\rho$  is the density (kg m<sup>-3</sup>) and *DM* is dry matter in % (w/w). As seen from the equations, dairy cattle slurry has a lower density and pig slurry a higher density at the same *DM* content. The flow properties for slurry have, in general, been found to be non-Newtonian at values of *DM* above 5%, while at lower *DM* values the slurry behaves as a Newtonian liquid (Landry et al., 2004). In general, animal slurry with a higher *DM* content shows pseudoplastic behavior, but for simple flow considerations an apparent viscosity will often suffice. Landry et al. (2004) correlated their viscosity data, measured for a *DM* between 9% and 14% at 20 °C, as follows:

Dairy cattle: 
$$\mu_{slurry} = 4.10^{-5} \cdot DM^{4.4671}$$
 (6)

Pig: 
$$\mu_{slurry} = 4.10^{-6} \cdot DM^{4.6432}$$
 (7)

where  $\mu_{slurry}$  is the apparent slurry viscosity (Pa s). Although these data should be used with caution, as Landry et al.'s (2004) comparison with other literature data shows, the conclusion is that pig slurry is less viscous than cattle slurry at a comparable *DM* content.

#### 5 Solid–Liquid Separation

Different techniques for separating slurry in a *DM*-rich and a liquid fraction have been developed and are used on farms. Solid–liquid separation may be carried out in settling tanks, where the solids are removed from the bottom of the tank, or the settling may be forced using centrifuges. Solids may also be removed mechanically by forced filtration using screw presses or drainage through fabric belts or screens. In-house separation can be viewed as an alternative to these solid–liquid separation techniques; however, this technology is not fully developed and is not used in practice. It will therefore not be described further in this review (see

Sect. 2). Numerous designs of solid–liquid separators exist for slurry mixed in the animal house, and the majority may be categorized as one of the four standard types of separators mentioned above.

When comparing results from different studies, it is an advantage to use one separation parameter expressing the efficiency of separation. The removal efficiency (R) expresses the efficiency of removal of a specific compound (x) from a slurry to the solid fraction. It is defined as follows:

$$R(x) = 1 - \frac{c(x)_{liquid}}{c(x)_{slurry}}$$
(8)

where  $c(x)_{slurry}$  and  $c(x)_{liquid}$  are the concentrations (g L<sup>-1</sup>) of the species in consideration (dry matter, P, N) in, respectively, the slurry being treated in the separator and the liquid fraction that is produced. The greater the removal efficiency, the lower the amount of compound *x* remaining in the liquid fraction. The equation characterizes the efficiency of the separator with respect to the liquid fraction; however, it does not give any indication of the production of the solid fraction.

A separation index  $(E_t)$ , on the other hand, expresses the distribution of the specific compound between the solid and liquid fractions:

$$E_t(x) = \frac{m_{x,solid}}{m_{x,slurry}} \tag{9}$$

where  $m_{x,slurry}$  and  $m_{x,solid}$  are the mass (g) of the compound in consideration in, respectively, the slurry being treated by the separator and of the solid fraction being produced. Hence, the values of both the liquid and the solid fraction are indicated. The larger the separation index, the greater the amount of compound x in the solid fraction.

The separation index, however, does not include information on the total amount of solid fraction relative to the total amount of liquid fraction. Thus, the simple separation index will theoretically be 50% if a machine separates the slurry into two equally sized fractions with similar contents of compound x. An improved expression for the separation may be obtained using the reduced separation index  $(E'_t)$ :

$$E'_{t}(x) = \frac{E_{t}(x) - \frac{m_{solid}}{m_{slurry}}}{1 - \frac{m_{solid}}{m_{slurry}}}$$
(10)

where  $m_{slurry}$  and  $m_{solid}$  are the total mass (g) of slurry being treated and the total mass of solids being produced.

As only approximately 25% of the studies presented in the literature on solid–liquid separators contain sufficient data to assess  $E_t$  and of these only two-thirds contain sufficient data to assess  $E_{t'}$ , the separation index ( $E_t$ ) has been used to compare efficiencies of separations in this review.

# 5.1 Sedimentation

#### 5.1.1 Technical Description

Sedimentation in a thickener is an attractive option for separation, due to the low cost and simple technology. Most thickeners consist of a container that is cylindrical at the top and conical at the bottom. In batch operation, slurry is added to the top of the thickener (Fig. 2) and the solids settle at the bottom of the conical part from whence the solids can be removed (Suzuki et al., 2002; Loughrin et al., 2006). To encourage settling and increase the transfer of solids settled on the upper part of the conical section, small thickeners can be vibrated while, for larger thickeners, this can be achieved by using a rake. Thickeners can also be operated in continuous mode, where slurry is added continuously while solid and liquid phases are removed at the same rate as slurry is added. In this case, the slurry has to be added in the separating zone (Fig. 2). Alternatively, lagoons may be used as settling basins.



Fig. 2 A typical thickener used for sedimentation

#### 5.1.2 Operation and Separation Efficiency

The time it takes for the solid to separate from the liquid phase can be estimated for dilute slurries from the terminal velocity of a single solid particle. The smallest particles will have the slowest terminal velocity and will therefore determine the final settling time. For small particles, the flow is laminar and the equation for the terminal velocity ( $v_{tg}$ ) simplifies to (Foust et al., 1980):

$$v_{tg}(d_p) = \frac{\left(\rho_{solid} - \rho_{liquid}\right) \cdot g \cdot d_p^2}{18 \cdot \mu_{slurry}} \tag{11}$$

where  $\mu_{slurry}$  is the viscosity of the slurry,  $\rho$  is the density of solid and liquid (see Sect. 4.5), g is the acceleration of gravity, and  $d_p$  is the diameter of the solid particle (all in SI units).

As seen, the settling velocity increases with the solid density compared with liquid density and particle size, but decreases with increasing slurry viscosity. As with solid density, because the viscosity and particle size vary from slurry to slurry, the terminal velocity can seldom be calculated in advance. At the same time, at least in the thickening zone of the thickener, the slurry cannot be assumed to be dilute and hindered settling will take place. Therefore, the terminal velocity as calculated in equation (11) alone cannot be used to determine the thickener size. Instead, the calculations or assessments have to be carried out based on laboratory settling experiments using the Coe-Clevenger method or the Talmadge-Fitch method, as described by Foust et al. (1980). However, the following general findings on slurry can be used as a yardstick.

Increasing the settling time increases the separation efficiency (Ndegwa et al., 2001; Converse and Karthikeyan, 2004). In a laboratory study, the settling of solids in pig slurry was studied for 4 hours. For slurry with an initial DM content of between 2% and 4%, the settling was observed to be completed within 1 hour. For pig slurry with a DM of 6% the settling time was over 4 hours, as was also observed for diluted slurries with 0.5% and 1% DM (Ndegwa et al., 2001). Interestingly, it was observed that the removal of DM was highest for 1% and 2% DM slurries (60% removal), while the DM removal was roughly 30% for the 0.5% and 4% slurries, and only 5% for the 6% DM slurry. The indications are that hindered settling and changes in slurry liquid properties slow down the settling velocity for pig slurries above 2% DM, while for very dilute slurries (i.e. 0.5% DM) the settling also slows down, presumably because at this low concentration fewer of the fine particles are co-precipitated with the larger faster-settling particles. Plant nutrients are not evenly distributed between particles of different density and size (see Sect. 4); in consequence, the settling of plant nutrients may not be linearly related to the settling of dry matter.

The mechanisms behind the slower settling velocities at higher DM seen in some studies are, for actual livestock slurries, due to a combination of hindered settling and increased density and viscosity of the slurry, as a large content of small particles will cause an increasing viscosity of the slurry liquid and thereby reduce the settling velocity (Henriksen et al., 1998a, b). It must be noted, though, that the effect of DM concentration on settling has not been observed in all studies (Gao et al., 1993). Also, the increased weight of the top sediment particles will squeeze water out of the thickening zone, causing turbulence that stirs up the particles; a mechanism that may be more pronounced in batch settling systems compared with technologies where the sediment is removed continuously (Foust et al., 1980). Fermentation and increased buoyancy of the particles due to the trapping of gas bubbles may reduce settling if the process is taking place over a long period of time, e.g. in lagoons; therefore it is recommended that slurry temperature should be kept below 16 °C (Meyer et al., 2007).

Cattle slurry has a higher viscosity than pig slurry at similar DM concentrations (see Sect. 4.5; Landry et al., 2004), most probably due to a larger amount of particles, which may contribute to a slower and less efficient settling of DM in cattle slurry. As for pig slurry, the DM settling of cattle slurry increases when DM concentration in the slurry increases; for example, from 0.1% to 1% (Moore et al., 1975). The settling rate of DM in cattle slurry decreases exponentially with time, and DM settling is almost at maximum after 1.5 hours (Moore et al., 1975). In contrast, because of the large P content in the small, slowly settling particles (Sect. 4.1), P settling may increase significantly over time and increase from about 50% settling after 4 hours to 75% settling after 48 days (Converse and Karthikeyan, 2004). On the other hand, the settling of total N may not increase with increased settling time (Converse and Karthikeyan, 2004). Most K and  $NH_4^+$ is dissolved in the liquid phase (see Sect. 4.3; Massé et al., 2007a); therefore most K and  $NH_4^+$  is recovered in the liquid phase after sedimentation of solids (Table 2; Massé et al., 2005).

Technical specifications	Slurry		Separatio	on index (%) <sup>2</sup>				
	01	Dry matter		D	NT / / 1	NTT N	D ( ) 1	D.C.
Sedimentation time (h)	Origin	(%)	Volume	Dry matter	N-total	NH <sub>4</sub> -N	P-total	Reference
Mean			22	56	33	28	52	
(standard deviation)			(4)	(10)	(2)	(2)	(21)	
0.2	Р	-	-	51	-	-	17	Powers and Flatow, 2002
NA	Р	2.0	18	63	-	-	65	Reimann, 1989
0.3	С	1.0	16	72	30	-	69	Sherman et al., 2000
4	С	1.3	25	52	35	27	42	Converse and Karthikeyan, 2004
24	С	2.5	25	42	33	28	46	Converse and Karthikeyan, 2004
1200	С	3.2	25	55	35	30	70	Converse and Karthikeyan, 2004

**Table 2** Separation indexes at sedimentation

<sup>1</sup> P = pig, C = cattle.

 $^{2}$  Separation index is the mass of a compound in the solid fraction compared to the mass of a compound in the original raw slurry.

## 5.2 Centrifugation

#### 5.2.1 Technical Description

Increasing the gravitational force can reduce the settling time needed to achieve a given separation efficiency. In practice, this is accomplished in decanter centrifuges, where a centrifugal force is generated to cause the separation.

There are vertical and horizontal types of decanter centrifuges. The horizontal decanter centrifuge (Fig. 3) uses a closed cylinder with a continuous turning motion. The centrifugal force separates solids and liquids at the wall into an inner layer with a high DM concentration and an outer layer consisting of a liquid containing a suspension of colloids, organic components and salts. The solid and liquid phases are transported to either end of the centrifuge by rotating the entire centrifuge at high speed and by simultaneously rotating the conveyor at a speed that differs slightly from the speed of the bowl (outer conical shell). The solid particles are conveyed towards the conical end and let out through the solid-discharge openings, whereas the supernatant flows towards the larger end of the cylinder formed by the bowl and the flights of the conveyor. During the transport of the slurry, the particles are separated from the liquid and the liquid phase is discharged through liquid-discharge openings at the wide end of the decanter centrifuge.

Fig. 3 A typical decanter centrifuge

#### 5.2.2 Operation and Separation Efficiency

For small particles in laminar flow, the terminal velocity ( $\nu_{tc}$ ) can be calculated as (Foust et al., 1980):

$$v_{tc}(d_p) = \frac{\left(\rho_{solid} - \rho_{liquid}\right) \cdot \omega^2 \cdot r \cdot d_p^2}{18 \cdot \mu_{slurry}}$$
(12)

where r is the distance of the particles from the centrifuge's axis of rotation and  $\omega$  is the angular velocity, all in SI units.

The only difference between the terminal velocity in a sedimentation tank and a centrifuge is the applied force; gravity versus centrifugal force. For simple laboratory centrifuges, the centrifuge efficiency can thus be related to the improvement in gravitational force, the *G* force (unit: g):

$$G = \frac{\omega^2 \cdot r}{g} \tag{13}$$

The efficiency of full-scale decanter centrifuges, however, cannot be described as simply as that. This is partly because the distance that the particles travel in a radial direction is large; thus the distance the particles have to travel from the center of the centrifuge, and therefore the settling velocity, vary during the sedimentation process; and partly because the geometry of decanter centrifuges is quite complicated.

The performance of a decanter centrifuge is often described by its feed handling capacity (Q), which can be calculated as (Foust et al., 1980):

$$Q = \frac{\left(\rho_{solid} - \rho_{liquid}\right) \cdot g \cdot d_{pc}^2}{18 \cdot \mu_{slurry}} \cdot 2 \cdot \Sigma = v_{t,g} \cdot 2 \cdot \Sigma$$
(14)

where Q is the volumetric feed rate,  $d_{pc}$  is the diameter of the smallest particle separated from the slurry in the centrifuge, and  $\Sigma$  is the sigma factor (m<sup>2</sup>), all in SI units.

The sigma factor,  $\Sigma$ , is a property of the specific centrifuge geometry and *G* force and can only be calculated in advance for simple laboratory centrifuges. For full-scale centrifuges,  $\Sigma$  has to be obtained from experiments. As  $v_{tg}$  is a function of the slurry alone, and  $\Sigma$  a function of the centrifuge alone, the sigma factor can be used to compare the efficiency of different decanter centrifuges.

The dewatering volume of a decanter is considered to be the total volume (V) of the liquid zone in



the cylindrical part of the drum. This volume may be changed by level regulators and the retention time (Rt)in seconds can be calculated as:

$$Rt = \frac{V}{Q} \tag{15}$$

where V is the dewatering volume of the decanter bowl  $(m^3)$ .

From equations (14, 15), it is obvious that the volumetric feed rate, and therefore the retention time, depends on the chosen value of  $d_{pc}$  (the smallest particle to be separated). Reducing the feed rate and thereby increasing the retention time automatically leads to better separation but less economical performance of the centrifuge.

Increasing the retention time by reducing the volumetric feed rate has been observed to increase the efficiency of the separation of slurry (Table 3; Sneath et al., 1988; Møller et al., 2007a). In consequence, the separation of DM and P was found to be high in the laboratory study of Vadas (2006) (see Table 3) using the relatively long retention time of 600 s.

Increasing the dewatering volume within the centrifuge increases the retention time (Eq. (15)) but reduces the thickening zone (conical water-free part). Therefore, increasing the dewatering zone will

increase the removal of DM from the liquid fraction. However, it will also reduce the drainage of water from the solid fraction (Reimann, 1989); hence the DMconcentration in the DM-rich fraction will decrease.

Increasing the angular velocity of the decanter centrifuge will increase the DM concentration of the solid fraction (Eq. (15)). Thus, using a velocity as high as 2560 g, the decanter centrifuge produced a solid fraction with a DM of 40% (Møller et al., 2007a). However, increasing the angular velocity had no effect on the separation of the P, K and N (Møller et al., 2007a).

The separation efficiency of DM increases at increasing DM content of the slurry (Table 3). The opposite result may have been expected, as seen in the sedimentation studies, because a higher viscosity of the slurry may reduce the settling velocity of the small particles (Reimann, 1989). It was hypothesized by Sneath et al. (1988) that vigorous stirring of the slurry may enhance the attachment of small particles to larger particles, and thereby improve settling of the small particles. This effect declines with dilution of the slurry; it has been observed that in the liquid from separation of a low-DM slurry, 70% of the particles were <4  $\mu$ m in liquid from separation of a high-DM slurry.

 Table 3
 Separation indexes at centrifugation

Technical specifications	Slurry		Separati	on index	$(\%)^2$			
CentrifugationCentrifugatiovelocity $(g)$ time $(s)$	n Origin <sup>1</sup>	Dry matter (%)	Volume	Dry matter	N-total	NH <sub>4</sub> -N	P-total	Reference
Mean			14	61	28	16	71	
(standard deviation)			(7)	(16)	(10)	(8)	(14)	
1500 600	Р	6.3	-	95	-	_	90	Vadas, 2006
2050 30	Р	8.9	22	69	34	22	87	Møller et al., 2007a
2050 8	Р	7.0	15	57	-	_	-	Sneath et al., 1988
2050 8	Р	4.2	8	52	-	_	_	Sneath et al., 1988
2050 8	Р	2.0	3	47	_	_	_	Sneath et al., 1988
2200 30	Р	5.1	9	51	17	10	71	Møller et al., 2007a
2200 30	Р	4.0	7	52	17	7	70	Møller et al., 2007a
2200 30	Р	6.8	26	70	36	26	82	Møller et al., 2007a
4100 600	Р	5.3	13	60	29	_	62	Møller et al., 2002
4100 600	Р	2.6	5	33	13	_	66	Møller et al., 2002
	Р	7.0	16	70	32	_	52	Reimann, 1989
1500 600	С	8.0		86	-	_	83	Vadas, 2006
2200 30	С	7.0	23	63	29	16	55	Møller et al., 2007a
4100 600	С	6.4	21	65	49	_	82	Møller et al., 2002
4100 600	С	4.5	12	55	27	_	79	Møller et al., 2002
	С	6.0	13	44	23	-	48	Reimann, 1989

<sup>1</sup> P = pig, C = cattle.

<sup>2</sup> Separation index is the mass of a compound in the solid fraction compared to the mass of a compound in the original raw slurry.

As a consequence of a higher *DM* content of cattle slurry compared with pig slurry, the *DM* separation index is higher for cattle slurry separation than for pig slurry separation (Table 3). For assessment of the retention of slurry, it is reasonable to assume that decanter centrifuges can retain particles >20–25  $\mu$ m in the *DM*-rich fraction (Sneath et al., 1988; Reimann, 1989; Møller et al., 2002).

Organic nitrogen and dissolved or adsorbed ammonium is related to the DM content of the solid fraction (see Sect. 3.1); therefore, total N separation is related to the DM content of the slurry being treated (Møller et al., 2002, 2007a; Sørensen and Møller, 2006). One should bear in mind that the relation between separation efficiencies and DM of treated slurry differs between centrifuges (Sneath et al., 1988; Møller et al., 2007a, b), and therefore an algorithm developed in one study cannot easily be applied to another. As decanter centrifuges do not retain the smallest particles, it is not surprising that no relation between separation of P and DM of treated slurry was observed (Møller et al., 2007a).

## 5.3 Drainage

#### 5.3.1 Technical Description

Many different kinds of slurry separation equipment that filter out solids from liquids are commercially available (Fig. 4). These use screens and filter belts to retain the solid fractions. With simple screens and



Fig. 4 A typical belt separator with pressure rolls

belt separators, the liquid is drained by gravity from solids in the separator. With a belt separator, the filter cake is continuously removed as the belt rotates, so that the raw-slurry loading area and solid-fraction unloading area change over and are cleaned continuously. Most of the tested filter separators are screens, which may have a variety of designs but often consist of a rotating perforated cylinder with a loading area at the top and a scraper to remove the solids. The liquid flows through the screen and is drained off.

#### 5.3.2 Operation and Separation Efficiency

The liquid flux through the filter is determined by the hydraulic resistance of the filter medium and the hydraulic resistance of the material deposited on the medium (i.e. the filter cake):

$$J = \frac{p}{\mu_{liquid} \left( R_m + SRF \, \frac{S \cdot V}{A} \right)} \tag{16}$$

where J is the flux (kg/m<sup>2</sup>/s),  $\mu_{liquid}$  is the viscosity of the permeate (Pa·s),  $R_m$  is the resistance to flow through the membrane (m<sup>-1</sup>), S is the particle concentration in feed (kg/m<sup>3</sup>), A is the filter area (m<sup>2</sup>) and V is the filtrate volume (m<sup>3</sup>). SRF is the specific filter cake resistance (m/kg), p is the pressure (Pa) and is related to the mass of the slurry (i.e.  $p = \rho gh$ ), and h is the height of suspension above the filter cake (m), which decreases during the process.

Particles can adhere to or clog the filter media and thereby increase the resistance of the filter (Massé et al., 2005), and substances such as hair in pig slurry may cause an immediate build-up of a filter cake with a low *SRF* (specific filter cake resistance). Thus, when the filter cake is formed, small particles often clog the filter pores (Severin and Grethlein, 1996). A mixture of particles with a particle size distribution between 1 and 100  $\mu$ m will produce a cake with a high *SRF*. This will reduce the liquid draining velocity from the filter cake and the effect will be a solid fraction with a low *DM* concentration (Karr and Keinath, 1978). Due to a higher fraction of larger particles (see Sect. 4.1), filter technology is more efficient in separating cattle slurry than pig slurry (Table 4).

As small particles are caught within the filter cake or adhere to the media, screens and filter will retain

Technical specificat	tions	Slurry		Separati	ion inde	$(\%)^2$			
Filter pore	Filtration		Dry matter		Dry				
size (mm)	time (min)	Origin <sup>1</sup>	(%)	Volume	matter	N-total	NH <sub>4</sub> -N	P-total	Reference
Mean				23	44	27	23	34	
(standard deviat	tion)			(16)	(27)	(17)	(19)	(21)	
0.1	(x)	Р	1.5-5.4	25	50	33	28	34	Holmberg et al., 1983
0.1	(4*x)	Р	1.5-5.4	45	67	51	47	59	Holmberg et al., 1983
0.11	120	Р	6.3	-	87	_	_	53	Vadas, 2006
0.5	_	Р	1.8	-	17	5	-	3	Pieters et al., 1999
0.8	_	Р	1.5	-	18	_	-	-	Hegg et al., 1981
2.5	(x)	Р	1.5-5.4	1	11	3	3	2	Holmberg et al., 1983
2.5	(4*x)	Р	1.5-5.4	13	23	16	15	23	Holmberg et al., 1983
3.0	_	Р	5.7	30	62	34	-	39	Møller et al., 2000
0.11	120	С	8.0	-	84	_	-	62	Vadas, 2006
0.8	_	С	1.1	-	13	_	-	-	Hegg et al., 1981
0.85	-	С	7.7	4	14	-	-	-	Gilbertson and Nienaber, 1978
3.0	_	С	7.1	44	56	49	_	49	Møller et al., 2000
_	_	С	6	19	54	22	_	20	Pain et al., 1978
-	-	С	12	30	58	29	_	29	Pain et al., 1978

Table 4 Separation indexes at drainage

<sup>1</sup> P = pig, C = cattle.

<sup>2</sup> Separation index is the mass of a compound in the solid fraction compared to the mass of a compound in the original raw slurry. x: unspecified constant.

not only particles larger than the size of the mesh or the screen openings but also smaller particles. Therefore, one cannot assess filtering efficiency simply from particle size distribution and plant nutrients in different particle fractions and the size of the screen openings.

Increasing the retention time of the filter cake on the screen or the filter fabric will increase the drainage, and hence the DM concentration of the solid fraction produced. Retention time is often longer in laboratory studies than in pilot and full-scale tests; therefore, laboratory studies often give the most efficient separation of DM and P (Table 4) (Vadas, 2006). The best strategy is to determine the *SRF* in the laboratory and use it for the design of pilot or full-scale equipment.

Most P is contained in the small particles or is dissolved in the liquid (Fig. 1), while little N is found on particles larger than 0.125 mm (see Sect. 4.2). Therefore, relatively little P and N is retained on screens or filter fabrics, which only retain large particles (Meyer et al., 2007). Nevertheless, some N, P and K are retained in the filter cake because the cake has a high water content (50–80%-volume is not uncommon) with dissolved N, P and K (Pain et al., 1978; Holmberg et al., 1983; Møller et al., 2000).

# 5.4 Pressurized Filtration

#### 5.4.1 Technical Description

The typical equipment used for filtration with applied pressure is a screw press or a press auger. In a press auger separator or a screw press separator, the effluent is transported into a cylindrical screen with a screw (Fig. 5). The liquid will pass through the screen and be collected in a container surrounding the screen. At the end of the axle the DM-rich fraction will be pressed against the plate and more liquid pressed out of the solid fraction. The solid phase will drop from the opening between the plate and the opening of the cylindrical mesh.

#### 5.4.2 Operation and Separation Efficiency

The liquid flux through the filter medium can be determined from equation (18) by setting p equal to the applied pressure. According to filtration theory, *SRF* (specific filter cake resistance) is constant during constant pressure filtration; however, for a



Fig. 5 A typical screw press

complex organic suspension such as slurry, *SRF* often increases during the process. The increasing *SRF* has been ascribed to sedimentation (Christensen and Dick, 1985), small particles blocking the pores in the cake (Sørensen et al., 1995), and a time-dependent compression of the cake (Christensen and Keiding, 2007). The slurry filter cake is compressed during pressure filtration; hence the *SRF* is several orders of magnitude higher for pressure filtration than for gravity drainage (Hjorth et al., 2008).

The cake compression ensures that the screw press can produce a solid fraction with a high DM content; often being twice as high as for gravity drainage (Møller et al., 2000). Increasing the applied pressure will increase the DM content of the solid fraction. Although aggregation of particles on the filter may, to some degree, contribute to the retention of small particles in the screw press, this has no significant effect, as the applied pressure forces small particles through the filter pores. A large proportion of small particles is therefore found in the liquid fraction after separation (Møller et al., 2002). Thus, the filter cake contains little N, P or K, because these are primarily found in the liquid phase and in the small particles which are drained off the filter cake with the permeate. In consequence, the plant nutrient separation efficiency of the screw press is low (Table 5).

Table 5 Separation indexes at pressurized filtration

Technical specifications		Slurry		Separat	ion inde	x (%) <sup>2</sup>			
Filter pore size (mm)	Filtration time (min)	Origin <sup>1</sup>	Dry matter (%)	Volume	Dry matter	N-total	NH <sub>4</sub> -N	P-total	Reference
Mean				11	37	15	-	17	
(standard deviation)				(15)	(18)	(17)	(-)	(14)	
0.7	_	Р	5.7	7	28	7	-	15	Møller et al., 2000
0.75	2	Р	6.3		64	_	_	46	Vadas, 2006
0.9	_	Р	5.7	5	28	6	_	12	Møller et al., 2000
1.0	_	Р	5.3	4	27	7	_	7	Møller et al., 2002
-	_	Р	1.8		51	31	_	42	Pieters et al., 1999
-	_	Р	1.8		26	11	_	7	Pieters et al., 1999
-	_	Р	6.3	7	21	4	_	13	Pos et al., 1984
0.75	2	С	8.0		64	_	_	28	Vadas, 2006
0.9	_	С	7.1	3	14	4	_	7	Møller et al., 2000
1.0	_	С	4.5	2	13	4	_	8	Møller et al., 2002
1	_	С	5.2	10	46	5	_	3	Wu, 2007
1.0	_	С	6.4	5	30	8	_	15	Møller et al., 2002
1.6	_	С	3.8	6	20	_	_	_	Pos et al., 1984
3.0	_	С	7.1	13	40	13	_	21	Møller et al., 2000
3.2	_	С	5.0	13	37	_	_	_	Pos et al., 1984
-	_	С	5.2	13	35	36	_	_	Pos et al., 1984
-	-	С	17.7	60	77	60	_	_	Menear and Smith, 1973

<sup>1</sup> P = pig, C = cattle.

<sup>2</sup> Separation index is the mass of a compound in the solid fraction compared to the mass of a compound in the original raw slurry.

# 5.5 Concluding Remarks about Slurry Separators

In general, centrifugation is the most efficient technique for the separation of DM and P (see Tables 2–5) and filtration is less efficient for separating DM, N and P. To some extent, the separation of  $NH_4^+$  follows the same pattern. The poorest N and P separation is achieved using pressurized filtration. The advantage of pressurized filtration, though, is the production of a solid fraction with a high DM concentration of the solid fraction. Therefore, drainage is sometimes followed by pressurized filtration to obtain both high separation efficiency and high DM concentration.

The choice of separators will depend on the objective of the separation. The goal may be to reduce *DM* content in the slurry to ease the transport of slurry in pipes or channels, reducing the risk of sedimentation or blockage of the pipes and channels, in which case simple screens or filters may be useful. This simple separation will also produce a liquid fraction that will contribute to a homogeneous spreading of plant nutrients on the field and reduced ammonia and odor emissions (Rubæk et al., 1996; Hansen et al., 2006). If the cost of separation has to be low and retention time is no problem, then sedimentation is a cheap technique that will reduce the plant nutrient composition of the slurry efficiently.

A screw press seems to be a good choice if the objective is to produce biomass with a high *DM* concentration suitable for incineration (Table 5). The filtration technologies may retain up to about one-quarter of the N and P in the slurry. In some cases this may be sufficient to achieve a harmonious balance between the amount of plant nutrient applied to the field and the plant demand on the livestock farm.

Of the above-mentioned solid-liquid separation techniques, the decanter centrifuge is the most efficient in retaining P and at the same time producing a DM-rich fraction low in water. In addition, this technique may produce a liquid fraction with a N:P:K ratio similar to the N:P:K requirements of the crop.

The produced solid and liquid fractions may be treated further to obtain valuable slurry products (see Sect. 7). The solid–liquid separation may be improved further by an initial addition of chemicals (see Sect. 6).

The major disadvantages of the above-mentioned solid–liquid separation technologies are that they require investment in expensive equipment, have a limited life span, and require maintenance, process control systems and an external power supply in order to run. These extra costs need to be balanced against the reduction in investment in and cost of slurry transportation between animal houses, slurry storage facilities and fields.

#### 6 Pre-treatment: Chemical Additions

Solid–liquid separation technologies often have a limited capacity to retain plant nutrients in the solid fraction and thereby produce a liquid fraction with a composition that fulfils the needs of the livestock and plant producer.

Additives such as brown coal, benthonite, zeolite, crystals, chitosan and efficient microorganisms are used by livestock farmers and have been examined in numerous studies (Kayser, 1997; Henriksen et al., 1998a; Sommer et al., 2005; Garcia et al., 2009); however, in this review, these additives will not be considered as a pre-treatment for solid–liquid separation, as reliable information about their efficiency is lacking.

The focus here is on examples of the most commonly used methods that have been developed and used as pre-treatments to sedimentation, centrifugation or filtration. This section concentrates on the theory and use of polymers and multivalent ions, as well as on struvite formation.

# 6.1 Precipitation, Coagulation and Flocculation

Flocculation, coagulation and precipitation are chemical pre-treatments that improve the mechanical solid–liquid separation of many suspensions (Gregory, 1973; Smith-Palmer et al., 1994; Ashmore and Hearn, 2000; Nowostawska et al., 2005). In most suspensions, colloidal particles will not aggregate because the particles are negatively charged and repel each other (Gregory, 1989). However, aggregation will be facilitated by adding (1) multivalent cations that cause coagulation and/or (2) polymers, whereby flocculation occurs. The addition of multivalent cations will also enhance the precipitation of P.

#### 6.1.1 Theory

Large amounts of phosphate will precipitate following the addition of multivalent cations to the slurry (Hjorth et al., 2008) due to formation of, for instance, FePO<sub>4</sub>,  $Fe_5(PO_4)_2(OH)_9$  and  $Ca_3(PO_4)_2$ .

At the same time, the multivalent cations cause coagulation of the particles in the slurry (Barrow et al., 1997; Sherman et al., 2000; Hjorth et al., 2008). At coagulation, the multivalent cations neutralize (or partially neutralize) a particle's negative surface charge by adsorbing the oppositely charged ions to the particle surface, creating a double layer (see Sect. 4.5) and thereby removing the electrostatic barrier that prevents aggregation; a process termed 'charge neutralization' (Fig. 6a). An optimum dose exists, and overdosing occurs when the adsorbed ions reverse the surface charge, thus counteracting aggregation (Gregory, 1989).

The addition of polyelectrolyte polymers to slurry induces flocculation. Polymer bridging is the main reaction mechanism, whereas patch flocculation is of limited significance, and charge neutralization is not important (Vanotti and Hunt, 1999; Hjorth et al., 2008). The addition of polymers will cause flocculation of particles and of existing but smaller aggregates that have been produced due to coagulation; for example, induced by the addition of Fe<sup>3+</sup> to the slurry.

Patch flocculation is the adsorption to particles of oppositely charged polyelectrolytes with a charge density much higher than the charge density of the particles. Thus, local positively and negatively charged areas are formed on the surface of the particles (Fig. 6b; Gregory, 1973). This results in a strong electrical attraction between the particles, especially when the electrical attraction extends far into the solution, i.e. at low conductivity (Gregory, 1973). A greater tendency to patch flocculation is expected for branched polymers compared with linear polymers, and for high-molecular-weight polymers (Bouyer et al., 2001; Bratskaya et al., 2005). However, since the conductivity of slurry is relatively high (see Sect. 4.5) patch flocculation is of minor relevance.

Polymer bridging occurs when long-chain polymers adsorb to the surface of more than one particle, causing the formation of strong aggregates of large flocs (Fig. 6c; Gregory, 1989). At high conductivity of the slurry or at high doses of added polymer (see Sect. 4.5), the polymer coils up and forms loops and tails. Due to steric hindrance between the particles, the loop and tail formation leads to deflocculation (Gregory, 1973). The maximum adsorbed mass of polymers increases with increasing ionic strength (Eriksson et al., 1993). Polymer characteristics of importance to polymer bridging are molecular weight and the molecular structure, as large molecules with long tails are capable of catching by-passing particles relatively easily. Charge density is also important, as this affects the number of loops and tails that are formed and the interaction between polymer and particles (Larsson et al., 1999; Ashmore and Hearn, 2000; Bouyer et al., 2001).



**Fig. 6** Diagrammatic representations of (a) coagulation, (b) patch flocculation and (c) polymer bridging (adapted from Hjorth et al., 2008)

#### 6.1.2 Technical Description

Multivalent ions and polymers need to be added carefully to the slurry in order to achieve satisfactory particle aggregation. If both additives are used, then the multivalent ion is added first to the slurry, which is then stirred to ensure homogeneous distribution of ions and dry matter (Fig. 7). Several minutes of slow stirring are necessary for the charge neutralization and coagulation to occur. Next, the polymer is slowly added in small doses during vigorous stirring, followed by slow stirring, which is necessary for polymer bridging and patch flocculation to occur. The shear applied (for example, by the impeller, i.e. time and stirring velocity), has a large impact on the formation of the aggregates; too low a shear causes the aggregates to be non-uniform and unstable with low particle catchment, while too large a shear causes the aggregates to be destroyed (Mikkelsen, 1999). After the addition and aggregation, the slurry may be transferred to ordinary solid-liquid separators (see Sect. 5; Fig. 7).



**Fig. 7** Example of the use of additives and separating the solids from the liquids using a filter bed separator with containers and stirrers for treatment with (1) slurry pumped into the separator, (2) coagulants added to the slurry pumped into the first container, and (3) polymers added to slurry transferred from the first to the second container. The additives may be added into the tubes transferring the slurry to the separator and between containers, or the additives may be added to the slurry in the containers that are equipped with a stirrer: vigorous stirring may break up the flocs

#### 6.1.3 Operation and Separation Efficiency

A broad range of multivalent cations intended to precipitate P and coagulate the slurry have been tested. These tests provide an opportunity to compare the efficiency of additives and to assess which multivalent ions provide the best slurry separation (Table 6). If the multivalent cation added to the slurry is calcium, the efficiency is ranked as CaO > Ca(OH)<sub>2</sub>; if the multivalent ion is iron, the efficiency is ranked as FeCl<sub>3</sub> > Fe<sub>2</sub>(SO<sub>4</sub>)<sub>3</sub> > FeSO<sub>4</sub>; while if the multivalent ion is aluminium, the efficiency is ranked as Al<sub>2</sub>(SO<sub>4</sub>)<sub>3</sub> > AlCl<sub>3</sub>.

The use of the separation products should also be considered when selecting the multivalent ion. For example, a decrease in pH is observed upon addition of FeCl<sub>3</sub> and Al<sub>2</sub>(SO<sub>4</sub>)<sub>3</sub> (Hjorth et al., 2009), which could be an advantage with respect to NH<sub>3</sub> emissions; while CaO causes pH to increase, which may be preferable when a nitrification/denitrification step is to follow the separation (Szögi et al., 2006). The environmental consequences of applying multivalent cations should also be considered (Nahm, 2005).

The literature reports on a wide range of polyacrylamide (PAM) polymers that have been examined, with characteristics that may vary with respect to charge, charge density, molecular size and structure (Table 7). These studies therefore provide an opportunity to compare polymers and to give an indication of the polymer characteristics likely to provide the best slurry separation.

Most studies indicate that a cationic polymer is superior to anionic and neutral polymers (Table 7), which correlates well with the fact that the particles in animal slurry are mainly negatively charged (see Sect. 4.5). Polymers of medium charge density (20-40 mol%) have been shown to be most efficient in most studies, confirming the hypothesis that polymer bridging is the main mechanism behind flocculation in slurry (Vanotti and Hunt, 1999; Hjorth et al., 2008). The reason is that a polymer of medium charge density has a large number of charged sites and hence would have the ability to catch particles efficiently; however, a polymer of medium charge density also has many non-charged sites and will therefore not neutralize the particles completely, thereby leaving charges available on the particle surface for another polymer. This is supported by the finding that optimal separation has been observed at approximately 30% neutralization of the

Slurry		Comparison			
origin <sup>1</sup>	ID	Precipitant/coagulant		Criteria <sup>2</sup>	Reference
Р	1	FeCl <sub>3</sub>	1 > 2 > 3 = 4	pH reduction	Hjorth et al., 2009
	2	$Fe_2(SO_4)_3$			
	3	$AICI_3$			
D	4	$AI_2(SO_4)_3$	1 0		
Р	1	$Fe_2(SO_4)_3$	1 > 2	Volume separation	Møller et al., 2007a
	2	$AI_2(SU_4)_3$	2 > 1	DM, N, P and NH4 separation	
Р	1	FeCl <sub>3</sub>	1 > 2	Liquid turbidity, volume separation	Estevez Rodríguez et al., 2005
	2	$Fe_2(SO_4)_3$	1 = 2	DM, VS, COD and P separation	
С	1	$Al_2(SO_4)_3$	1 > 2, 3	pH reduction	Karthikeyan et al., 2005
	2	FeCl <sub>3</sub>	1 = 2 > 3	DM, P, N, NH <sub>4</sub> and TOC separation	
	3	CaO			
С	1	AlCl <sub>3</sub>	1 > 2	pH reduction	Krumpelman et al., 2005
	2	FeCl <sub>3</sub>	2 > 1	P separation	
			1 = 2	DM separation	
Р	1	$Al_2(SO_4)_3$	1 = 2 = 3 > 4 = 5	DM separation	Powers and Flatow, 2002
	2	FeCl <sub>3</sub>	1 = 2 > 3 > 4 = 5	Volume separation	
	3	CaO	1 = 2 > 3 > 4 > 5	P separation	
	4	FeSO <sub>4</sub>			
C	5		1 0		D 1D 11 0000
C	1	$Al_2(SO_4)_3$ E <sub>2</sub> Cl	l = 2	DM separation	Dao and Daniel, 2002
D	2		2 > 1	pri reduction	71 1.1 1000
Р	1 2	$Al_2(SO_4)_3$ FeCl <sub>3</sub>	1 = 2	DM separation	Zhang and Lei, 1998
Р	1	FeCl <sub>3</sub>	1 > 2	DM separation	Ndegwa et al., 2001
	2	$Al_2(SO_4)_3$	2 > 1	P separation	
С	1	FeCl <sub>3</sub>	1 > 2	Volume, P and N separation	Sherman et al., 2000
	2	$Al_2(SO_4)_3$	2 > 1	DM separation	
С	1	FeCl <sub>3</sub>	1 = 2 = 3	DM, P separation	DeBusk et al., 2008
	2	$Al_2(SO_4)_3$			
	3	AlCl <sub>3</sub>			
Р	1	FeCl <sub>3</sub>	1 > 2 = 3	DM separation	Gao et al., 1993
	2	Ca(OH) <sub>2</sub>			
	3	$Al_2(SO_4)_3$			
С	1	CaO	1 > 2 > 3	DM separation	Barrow et al., 1997
	2	Ca(OH) <sub>2</sub>	1 = 2 > 3	pH increase	
	3	$CaSO_4$			
С	1	$Fe_2(SO_4)_3$	1 > 2	DM, N and P separation	Barrow et al., 1997
	2	FeSO <sub>4</sub>			
С	1	FeCl <sub>3</sub>	1 > 2	DM, N and P separation	Barrow et al., 1997
	2	$Fe_2(SO_4)_3$			

Table 6 Comparison of precipitants/coagulants with respect to separation-relevant criteria

<sup>1</sup> P = pig, C = cattle.

 $^{2}DM = dry$  matter, COD = chemical oxygen demand, TOC = total organic carbon, VS: volatile solids.

particle surface charge (Hjorth et al., 2008). Further indications are that linear polymers are preferable to branched polymers, and polymers of large molecular

weight are preferable to those of smaller molecular weight (Table 7). The reason seems to be the very efficient capture of smaller particles by the large, loose

origin lIdTrade nameChargeCharge density (moP1Superfloc c-2260+402Zetag 7878FS40+802Superfloc c-2230+103Superfloc c-2230+105Superfloc c-2260+206MAP+207MAP+207MAP-208MAP-3491Magnafloc 90L-1Magnafloc 90L-342Magnafloc 120L-3491NA+407-3Nad91NA+2091NA+2091NA+7091NA+7091NA+7091Nordfloc CE 169+609Nordfloc CE 169+603Nordfloc CE 169+60	_	Flocculant (polymer)							
$ \begin{array}{c cccccc} \hline P & 1 & Superfloc c-2260 & + & 40 \\ \hline 2 & Zetag 7878FS40 & + & 80 \\ \hline 2 & Superfloc c-2230 & + & 15 \\ \hline 3 & Superfloc c-2230 & + & 16 \\ \hline 3 & Superfloc c-2240 & + & 20 \\ \hline 5 & Superfloc c-2250 & + & 20 \\ \hline 5 & Superfloc c-2250 & + & 40 \\ \hline 7 & A & + & 20 \\ \hline 7 & A & + & 13 \\ \hline 7 & A & A & + & 13 \\ \hline 7 & A & A & + & 13 \\ \hline 7 & A & A & + & 13 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 20 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 7 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 8 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & + & 40 \\ \hline 9 & A & A & A & + & 40 \\ \hline 9 & A & A & A & + & 40 \\ \hline 9 & A & A & A & + $	Id	Trade name	Charge	Charge density (mol%)	Structure	Molecular weight	Comparison	Criteria <sup>2</sup>	Reference
P       I       Superfloc c-2220       +       80         P       1       Superfloc c-2230       +       10         2       Superfloc c-2240       +       20         3       Superfloc c-2250       +       20         5       Superfloc c-2250       +       20         6       Superfloc c-2250       +       20         7       4       McP1       +       20         7       3       MAP       -       40         7       4       20       -       34         8       Magnafloc 90L       -       10       -         7       3       Magnafloc 120L       -       34         8       SCP       +       13       -         9       1       NA       +       7         7       -       34       -       -         8       Magnafloc 120L       -       34       -         7       -       -       10       -       -         8       NA       +       -       -       -         9       NA       -       -       -       -         1	-	Superfloc c-2260	+	40	Linear	Large		P separation, Liquid turbidity	Hjorth et al., 2008
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$							1 > 2	Dewatering at drainage	
P       1       Superfloc c-2220       +       10         2       Superfloc c-2240       +       20         3       Superfloc c-2240       +       20         5       Superfloc c-2250       +       20         6       Superfloc c-2260       +       20         7       MCP1       +       Medium         8       MAP       -       40         7       3       MAP       -       40         8       SCP       +       Medium         9       1       Magnafloc 90L       -       34         7       3       Zetag 7837       +       10         8       SCP       +       10       34         9       1       NA       +       7         7       3       Zetag 7837       +       13         8       NA       +       7       34         9       1       NA       +       7         7       3       NA       +       7         8       No       +       4       4         7       3       Nordfloc CE 169 WD       +       70	5	Zetag 7878FS40	+	80	Branched	Medium	1 = 2 1 < 2	DM separation Dewatering at pressurized filtration	
2       Superfloc c-2230       +       15         3       Superfloc c-2240       +       20         5       Superfloc c-2260       +       20         5       Superfloc c-2260       +       20         5       Superfloc c-2260       +       20         7       MAP       +       Medium         3       MAP       -       Medium         4       SCP       +       Medium         3       MAP       -       10         4       SCP       +       10         2       Magnafloc 90L       -       34         3       Zetag 7867       +       10         2       NA       +       7         3       XA       +       4         2       NA       +       13         3       NA       +       7         3       NA       +       7         6       NA       +       7         7       +       13       13         8       No       +       4         9       No       +       70         6       N       +	-	Superfloc c-2220	+	10	Linear	Large	5 > 1, 2, 3, 4	Liquid turbidity	Hjorth et al., 2009
3       Superfloc c-2240       +       20         5       Superfloc c-2250       +       20         5       Superfloc c-2250       +       20         5       Superfloc c-2250       +       28         7       Map       +       28         8       MAP       +       28         9       MAP       -       Medium         1       Magnafloc 90L       -       10         2       Magnafloc 90L       -       34         2       Magnafloc 120L       -       34         3       Zetag 7867       +       4         7       -       34       -         8       I       NA       +       7         9       I       NA       +       7         7       -       34       -       -         8       NA       +       7       -         9       NA       +       7       -         8       No       +       4       -         9       NA       +       7       -         9       Nordfloc CE 169       +       70       -	2	Superfloc c-2230	+	15	Linear	Large		•	
4       Superfloc c-2250       +       28         5       Superfloc c-2260       +       40         2       MCP1       +       Medium         3       MAP       -       40         4       SCP       +       Medium         3       MAP       -       Medium         4       SCP       +       Large         3       Magnafloc 90L       -       34         2       Magnafloc 90L       -       34         2       Magnafloc 120L       -       34         3       Zetag 7867       +       10         2       NA       +       7         3       XA       +       4         2       NA       +       7         3       NA       +       4         4       NA       +       7         7       +       13       13         8       NA       +       7         8       NA       +       7         9       NA       +       70         6       NA       +       70         7       1       1       13	3	Superfloc c-2240	+	20	Linear	Large			
5       Superfloc c-2260       +       40         2       MCP1       +       Medium         3       MAP       -       +       Medium         3       MAP       -       Medium         4       SCP       +       10         2       Magnafice 90L       -       10         2       Magnafice 90L       -       34         2       Magnafice 120L       -       34         3       Zetag 7867       +       13         4       Zetag 7867       +       20         9       1       NA       +       4         2       NA       +       20         8       NA       +       7         9       NA       +       7         7       +       20       +         8       NA       +       7         8       NA       +       7         8       Nordfloc CE 169       +       70         9       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60         4       10       +       60 <td>4</td> <td>Superfloc c-2250</td> <td>+</td> <td>28</td> <td>Linear</td> <td>Large</td> <td></td> <td></td> <td></td>	4	Superfloc c-2250	+	28	Linear	Large			
C 1 MCP1 + Medium 2 MCP2 + Medium 3 MAP - Medium 4 SCP - Medium 4 SCP - Medium 2 Magnafloc 90L - 10 2 Magnafloc 120L - 34 4 Zetag 7867 + 13 2 NA + 20 P 1 NA + 20 3 NA + 20 6 NA + 20 7 7 7 7 8 NA + 20 8 NA + 70 9 Nordfloc CE 169 WD + 60 7 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NOrdfloc CE 169 WD + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1 NO + 60 1	2	Superfloc c-2260	+	40	Linear	Large			
2       MCP2       +       Medium         3       MAP       -       Medium         4       SCP       +       Large         4       SCP       -       10         2       Magnafloc JOL       -       10         3       Zetag 7837       +       13         4       Zetag 7867       +       20         9       1       NA       +       4         7       +       20       34         8       Zetag 7867       +       4         7       +       20       34         8       NA       +       7         9       NA       +       7         7       +       20       4         8       NA       +       70         8       NA       +       70         9       Nordfloc CE 169 WD       +       60         3       Nordfloc CE 169 WD       +       60         3       Nordfloc CE 169 WD       +       60	1	MCP1	+	Medium	I	I	1 > 2 > 4 > 3	DM, VS and COD separation	Rico et al., 2007
3       MAP       -       Medium         4       SCP       +       Large         2       Magnaficc 90L       -       10         2       Magnaficc 120L       -       34         3       Zetag 7867       +       13         4       Zetag 7867       +       20         9       1       NA       +       20         7       +       20       3       8         3       NA       +       4       4         4       NA       +       7       7         3       NA       +       7       7         6       NA       +       7       7         7       +       20       4       40         6       NA       +       70       7         7       +       20       +       70         8       1       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60	2	MCP2	+	Medium	I	1		4	
P       1       Magnaficc 90L       +       Large         2       Magnaficc 90L       -       10         2       Magnaficc 120L       -       34         3       Zetag 7867       +       13         4       Zetag 7867       +       20         P       1       NA       +       4         2       NA       +       20         3       NA       +       20         4       NA       +       7         5       Magnifloc C-1596       +       40         6       NA       +       70         8       1       Nordfloc CE 169 WD       +       60         3       Nordfloc CE 169 WD       +       60         3       Nordfloc CE 169 WD       +       60	1	MAP	1	Medium	Ι	I			
P       1       Magnafloc 90L       -       10         2       Magnafloc 120L       -       34         3       Zetag 7867       +       13         4       Zetag 7867       +       20         9       1       NA       +       4         2       NA       +       20         3       NA       +       7         4       NA       +       13         5       Magnifloc C-1596       +       20         6       NA       +       70         7       3       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60	4	SCP	+	Large	I	I			
2       Magnafloc 120L       -       34         3       Zetag 7837       +       13         4       Zetag 7867       +       20         4       Zetag 7867       +       20         2       NA       +       4         3       NA       +       7         4       NA       +       7         3       NA       +       20         6       NA       +       20         7       13       3       10         8       NA       +       20         6       NA       +       20         7       13       13       13         8       Nordfloc C-1596       +       70         9       Nordfloc CE 169       +       60         2       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60	1	Magnafloc 90L	I	10	I	I	1 = 2 > 3 = 4	DM separation	Szögi et al., 2006
B       Zetag 7837       +       13         4       Zetag 7867       +       13         4       Zetag 7867       +       20         2       NA       +       4         2       NA       +       7         3       NA       +       7         4       NA       +       7         5       Magnifloc C-1596       +       40         6       NA       +       70         9       I       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60	2	Magnafloc 120L	I	34	I	I		Dewatering at	) )
3       Zetag 7837       +       13         4       Zetag 7867       +       13         4       Zetag 7867       +       20         2       NA       +       4         2       NA       +       7         3       NA       +       7         4       NA       +       7         5       Magnifloc C-1596       +       20         6       NA       +       70         9       1       Nordfloc CE 169       +       60         2       Nordfloc CE 169       +       60       +       40	1			-				filtration/pressure	
P       1       NA       +       20         P       1       NA       +       4         2       NA       +       7         3       NA       +       7         4       NA       +       13         5       Magnifloc C-1596       +       20         6       NA       +       70         9       1       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60	ŝ	Zetag 7837	+	13	Ι	I			
P       1       NA       +       4         2       NA       +       7         3       NA       +       13         4       NA       +       20         5       Magnifloc C-1596       +       40         6       NA       +       70         P       1       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60         3       Nordfloc CE 169       +       60	4	Zetag 7867	+	20	I	I			
2 NA + 7 3 NA + 13 4 NA + 20 5 Magnifloc C-1596 + 40 6 NA + 70 1 Nordfloc CE 169 + 60 2 Nordfloc CE 169 WD + 60 3 Nordfloc CE 129 + 40	1	NA	+	4	I	I	4 = 5 > 1, 2, 3, 6	DM separation	Vanotti et al., 2005
3       NA       +       13         4       NA       +       20         5       Magnifloc C-1596       +       40         6       NA       +       70         P       1       Nordfloc CE 169       +       60         2       Nordfloc CE 169       +       60         3       Nordfloc CE 129       +       40	2	NA	+	7	Ι	I	5 > 1, 2, 3, 4, 6	Dewatering at	
<ul> <li>3 NA</li> <li>4 NA</li> <li>5 Magnifloc C-1596</li> <li>6 NA</li> <li>70</li> <li>6 NA</li> <li>70</li> <li>71</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> <li>70</li> &lt;</ul>								filtration/pressure	
4 NA 5 Magnifloc C-1596 + 20 6 NA 1 Nordfloc CE 169 + 60 2 Nordfloc CE 169 WD + 60 3 Nordfloc CE 169 WD + 60	3	NA	+	13	Ι	I			
5 Magnifloc C-1596 + 40 6 NA + 70 2 Nordfloc CE 169 + 60 2 Nordfloc CE 169 WD + 60 3 Nordfloc CE 129 + 40	4	NA	+	20	I	I			
P         I         Nordfloc CE 169         +         70           2         Nordfloc CE 169 WD         +         60           3         Nordfloc CE 129         +         40	5	Magnifloc C-1596	+	40	I	Large			
P         1         Nordfloc CE 169         +         60           2         Nordfloc CE 169 WD         +         60           3         Nordfloc CE 129         +         40	6 1	NA	+	70	I	)			
2 Nordfloc CE 169 WD + 60 3 Nordfloc CE 129 + 40	1	Nordfloc CE 169	+	60	Linear	Large	1, 2, 3 > 4	Dewatering at	Estevez Rodríguez
3 Nordfloc CE 129 + 40	2	Nordfloc CE 169 WD	+	09	Linear	Large	1 = 2 = 3	Initration/pressure DM, VS, COD and P senaration	et al., 2005
	3	Nordfloc CE 129	+	40	Linear	Large			
4 Nordfloc CE 166 TR + 60	4	Nordfloc CE 166 TR	+	09	Branched	Large			

Table 7	(con	ttinued)							
Slurry		Flocculant (polymer)							
origin <sup>1</sup>	Id	Trade name	Charge	Charge density (mol%)	Structure	Molecular weight	Comparison	Criteria <sup>2</sup>	Reference
C	-	Superfloc A 1885 RS	I	Large	I	1	5,6,7,8>1,2,3,4	DM and P separation	Timby et al., 2004
	0	Superfloc A 1883	I	Medium	I	I	1 = 2 = 3 = 4 = 5 $= 6 = 7 = 8$	Soluble reactive P senaration	
	б	Superfloc A 1849 RS	I	Small	I	I	) - )		
	4	Superfloc N 1986	0	0	I	1			
	2	Superfloc C 1594	+	Medium	I	I			
	9	Magnifloc 240 G	+	Medium-large	I	I			
	2	Superfloc C 1598	+	Large	I	I			
	×	Magnifloc 255 G	+	Large					
U	1	C-1596	+	40	I	Large	1 = 2 = 3	DM seperation	Dao and Daniel, 2002
	0	C-581	+	100	I	Large		×	×
	б	SD-2085	+	55	I	Medium			
Ь	1	Magnifloc 494 C	+	20	I	I		Volume. DM. COD.	Vanotti and Hunt. 1999
		0						organic-N and	
								organic-P separation	
	7	Magnifloc 985 N	0	0	I	-1 > 2 = 3		NH4 and ortho-P seneration	
	б	Magnifloc 844 N	I	33	Ι	- 1=2=3		and a second second second second second second second second second second second second second second second	
Ь	1	Magnifloc 494 C	+	20	I	I	1>2>3>4	DM seperation	Garcia et al., 2007
	~	Excel I Iltra 5000	- +	75	I	I	1 > 2 = 3 > 4	COD seneration	
	1 ന	Magnifloc 120L	- 1	34	I	I	1 > 3 > 2 = 4	P seperation	
	4	Magnifloc 985N	0	0	I	I		J	
Ь	1	Excel 5020	+	20	I	I	1 > 2 > 3 = 4	Volume and DM	Vanotti and Hunt, 1999
								seperation	
	0	Excel 5020	+	40	I	I			
	З	Excel 5055	+	55	I	I			
	4	Excel 5000	+	75	I	I			
P and C	1	Magnifloc 240 G	+	Large	Ι	Large	4 > 1, 2, 3, 5	DM seperation	Zhang and Lei, 1998
	0	Magnifloc 240 GL	+	Large	I	Large			
	Э	Magnifloc 240 GSD	+	Large	Ι	Large			
	4	Magnifloc 255 G	+	Large	I	Large			
	S	Magnifloc 1596 C	+	Large	Ι	Large			
C	-	P 234 GD	+	I	I	I	1 = 2	DM, N and P seperation	Sherman et al., 2000
	7	PC 494	+	I	I	I			
<sup>1</sup> P = pi <sup>2</sup> DM =	ig, C : = dry	= cattle. matter, COD = chemic:	al oxygen	ı demand.					

flocs and/or that the shape of large, loose flocs is an advantage in the solid-liquid separators used to treat the slurry, especially when using separators where liquid is drained off the solids.

The separation products are often destined to be deposited in landfills or applied to cultivated fields; thus the environmental and health consequences of the applied polymer must be considered. The monomers of PAM used in most slurry separation studies (Table 7) can be toxic; however, a study on separated slurry products showed the risk to be minimal (Schechter et al., 1995). In the USA, these additives are considered to be 'generally recognized as safe' (GRAS) when added to the slurry below a specific application rate related to the end-use (Vanotti et al., 2002). However, there is a need for further studies on the efficiency of alternative polymer types for slurry separation, with the objective of replacing PAM with potentially less toxic polymers. Furthermore, when considering new polymers, one must take into consideration the toxicity of the organic components produced during degradation of the polymers.

The chemical and physical properties of slurry may vary (see Sect. 2) because slurry varies between livestock farms. During a separation run, slurry properties may vary even if the slurry is stirred when treated. Consequently, the optimal application rate of multivalent cations and polymers to slurry varies between farms and during a separation run. Various analytical methods have proved useful for evaluating the efficiency of different additives (Table 8). Of these, measurement of the DM content of the solids, dewatering ability, surface erosion from flocs, turbidity of the liquid fraction, capillary suction time, viscosity of the liquid fraction and floc size have proved to be useful as laboratory tests of slurry separation (Hjorth and Christensen, 2008).

The separation technique affects the separation efficiency of slurry treated with multivalent ions and flocculants. As for the combined solid-liquid separation technique, drainage appears to be superior when considering P and DM separation (Table 9), as pressure may disintegrate the flocs.

The addition of a coagulant/precipitant (a multivalent cation) increases the P separation (Table 9). There is a maximum multivalent ion amount above which no additional P will be retained in the solid fraction. The maximum separation index shows that 99% of the P in treated slurry is retained in the solid fraction (Sherman et al., 2000; Estevez Rodríguez et al., 2005). Due to coagulation, the addition of the multivalent cation may increase the DM and N separation (Barrow et al., 1997; Sherman et al., 2000). In theory, the multivalent ion may be overdosed when all dissolved P has

Table 8 Slurry characteristics potentially useful for assessing optimal coagulant and polymer dosage to slurry prior to separation

	Observed sample		
Slurry characteristics	Flocculated sample	Liquid fraction	Solid fraction
DM content		Х	Х
VS content		Х	Х
P content		Х	Х
N content		Х	Х
NH <sub>4</sub> content		Х	Х
Metal content		Х	Х
Dewaterability	Х		
Settling	Х		
Turbidity		Х	
Viscosity		Х	
Surface erosion from flocs	Х		
Capillary suction time (CST)	Х		
Floc size	Х		
Particle dispersion ánalysis (PDA)	Х		
Streaming current	Х		
VS: volatile solids			

VS: volatile solids.

Technical description			Slurry		Separation	index (%)	2			
Separation technique	Coagulant/ precipitant	Flocculant	Origin <sup>1</sup>	Dry matter (%)	Volume	Dry matter	N-total	NH4-N	P-total	Reference
Mean					22	70	43	20	79	
(standard deviation)					16)	(13)	(24)	(14)	(21)	
Sedimentation	$Al_2(SO_4)_3$	(none)	Ь	I	I	78	I		62	Powers and Flatow, 2002
Sedimentation	FeC1 <sub>3</sub>	(none)	Ь	I	I	82	I	I	64	Powers and Flatow, 2002
Sedimentation	$Fe_2(SO_4)_3$	(none)	Р	I	Ι	58	I	Ι	38	Powers and Flatow, 2002
Sedimentation	$CaCO_3$	(none)	Ь	I	I	61	I	I	18	Powers and Flatow, 2002
Sedimentation	CaO	(none)	Р	I	I	75	I	I	50	Powers and Flatow, 2002
Centrifugation	CaCl <sub>2</sub>	(none)	Р	3.1	10	34	1	10	59	Poels et al., 1988
Centrifugation	$CaCl_2$	(none)	Р	3.3	10	64	28	21	86	Poels et al., 1988
Centrifugation	(none)	Optifloc c-6260	Р	1.2	5	55	20	8	72	Møller et al., 2007a
Centrifugation	(none)	Optifloc c-6260	Р	2.8	13	84	54	18	85	Møller et al., 2007a
Centrifugation	$Fe_2(SO_4)_3$	Optifloc c-6260	Р	1.2	9	56	24	6	75	Møller et al., 2007a
Centrifugation	$Fe_2(SO_4)_3$	Optifloc c-6260	Р	2.1	5	54	16	L	73	Møller et al., 2007a
Centrifugation	$Al_2(SO_4)_3$	Optifloc c-6260	Р	2.1	10	73	35	15	91	Møller et al., 2007a
Centrifugation	$Al_2(SO_4)_3$	Optifloc c-6260	Р	4.2	18	87	45	28	90	Møller et al., 2007a
Drainage	(none)	Superfloc c-2260	Р	7.0	35	79	45	36	85	Hjorth et al., 2009
Drainage	FeC1 <sub>3</sub>	Superfloc c-2260	Р	7.0	43	88	62	50	76	Hjorth et al., 2009
Drainage	FeC1 <sub>3</sub>	Nordfloc CE 169	Р	2.5	I	70	I	I	66	Estevez Rodríguez et al., 2005
Drainage	FeC1 <sub>3</sub>	Nordfloc CE 169WD	Р	2.5	I	65	I	I	66	Estevez Rodríguez et al., 2005
Drainage	FeC1 <sub>3</sub>	Nordfloc CE 129	Р	2.5	I	70	I	I	66	Estevez Rodríguez et al., 2005
Drainage	$Fe_2(SO_4)_3$	Nordfloc CE 169	Ρ	2.5	I	70	I	I	66	Estevez Rodríguez et al., 2005
Drainage	$Fe_2(SO_4)_3$	Nordfloc CE 169WD	Р	2.5	I	67	I	I	66	Estevez Rodríguez et al., 2005
Drainage	$\operatorname{Fe}_2(\mathrm{SO}_4)_3$	Nordfloc CE 129	Р	2.5	I	65	I	I	66	Estevez Rodríguez et al., 2005
Sedimentation	$Al_2(SO_4)_3$	(none)	C	1.1	54	92	83	I	66	Sherman et al., 2000
Sedimentation	FeC1 <sub>3</sub>	(none)	С	1.1	47	78	75	I	94	Sherman et al., 2000
Sedimentation	(none)	P-234GD	C	1.1	29	99	35	I	67	Sherman et al., 2000
Sedimentation	(none)	P C-494	C	1.1	29	67	36	I	67	Sherman et al., 2000
Drainage	AlCl <sub>3</sub>	(none)	C	2.0	I	60	I	I	I	Timby et al., 2004
Drainage	AIC1 <sub>3</sub>	(undefine)	C	2.6	I	76	80	I	I	Worley et al., 2008
Pressurized filtration	CaO	MCP1	С	5.0	24	69	48	I	78	Rico et al., 2007
<sup>1</sup> P = pig, C = cattle. <sup>2</sup> Separation index is th	e mass of a con	npound in the solid fractio	n compared	to the mass of	a compoune	d in the ori	ginal raw s	lurry.		

been precipitated and all particles have been charge neutralized. Overdosing will cause the separation indexes to decrease, because the particles then become positively charged and repel each other.

Polymer (flocculant) addition improves DM, N and P separation (Table 9), because floc formation increases the amount of DM retained in the solid fraction and also the N- and P-containing organic and inorganic particles. Depending on the polymer, the structure of the flocs varies. The addition of linear polymers of large molecular weight produces large, loose flocs; while the addition of branched polymers of small molecular weight produces small, dense flocs (Hjorth et al., 2008). As sedimentation and centrifugation efficiency depend on particle density, it may be advantageous to produce small, dense flocs using branched small-molecular-weight polymers. Alternatively, when using filtration without applied pressure, draining of the liquid should be eased by producing large, loose flocs, resulting in a filter cake with high porosity. Filtration with applied pressure is favored by non-compressible flocs; and thus small, dense flocs may be the floc structure giving the best separation.

Ammonium and K are dissolved in the liquid (see Sect. 4.3); hence the polymer will not improve the  $NH_4^+$  or K separation. Due to the positive charges of  $NH_4^+$  and K, a multivalent cation will not improve the separation through precipitation or charge neutralization. Adding polymers may increase  $NH_4^+$  and K content in the solid fraction (Hjorth et al., 2009), because liquid with dissolved  $NH_4^+$  and K is retained in the flocs, as floc liquid is difficult to remove by mechanical solid–liquid separation (Vesilind, 1994).

The charge of DM particles have to be neutralized. Therefore at increasing charge of the DM and increasing concentration of the particles demands increasing addition of multivalent cation and polymer to a successful separation (Zhang and Lei, 1998). In contrast, decreasing amounts of polymer are needed when coagulation is improved by pre-treatment with an increased amount of multivalent cations (Krumpelman et al., 2005; Hjorth et al., 2009).

The separation indexes of untreated slurry (Tables 2–5) are much lower than the indexes of slurry treated with flocculants and coagulants (Table 9), with flocculation having a larger impact on filtration than on centrifugation (Hjorth et al., 2008). Thus, separation is improved by flocculation and further improved when multivalent ions are also added to the slurry.

## 6.2 Struvite Crystallization

Slurry contains struvite or the inorganic components contributing to the formation of struvite crystals. Thus, physical and chemical changes in slurry may initiate struvite formation. This is a problem in biogas plants because struvite will build up on pipe surfaces and reduce the capacity of slurry transport, but struvite formation may also be used to remove P and  $NH_4^+$  from the slurry.

#### 6.2.1 Theory

Crystallization of phosphate is used to facilitate the sedimentation of phosphorus as particles. Animal slurry contains P in the form of struvite (MgNH<sub>4</sub>PO<sub>4</sub>  $\cdot$  6H<sub>2</sub>O) and apatite Ca<sub>5</sub>OH(PO<sub>4</sub>)<sub>3</sub> (hydroxyapatite) crystals (see Sect. 4; Bril and Salomons, 1990; Greaves et al., 1999; Suzuki et al., 2002). Struvite has been shown to be the most significant of the solid forms of P in slurry, and struvite crystallization has been widely used as a method for separating P from wastewater and slurry (Ohlinger et al., 2000; Burton and Turner, 2003). Hydroxyapatite is not as attractive as a means of removing P from slurry and wastewater, because much calcium will have crystallized as calcite CaCO<sub>3</sub> in slurry.

$$Mg^{2+} + NH_4^+ + PO_4^{3-} + 6H_2O$$
  

$$\leftrightarrow MgNH_4PO_4 \cdot 6H_2O(s) \qquad (17)$$

The formation constant (K<sub>0</sub>) of struvite is  $1.41 \times 10^{13}$ and the equilibrium ion-activity product (IAPeq) is  $7.08 \times 10^{-14}$ . In slurry, the product of [Mg<sup>2+</sup>], [NH<sub>4</sub><sup>+</sup>] and  $[PO_4^{3-}]$  is generally lower than the conditional formation constant, due to low concentrations of dissolved magnesium (Mg<sup>2+</sup>) (Sommer and Husted, 1995a; Nelson et al., 2003), so little or no struvite is formed. The dissolved Mg may be lower than that determined with traditional extraction and measuring techniques, because a major fraction of the Mg<sup>2+</sup> ions may form complexes with dissolved organic matter (Bril and Salomons, 1990; Christensen et al., 2009). The concentration of  $PO_4^{3-}$  will also have a major impact on struvite crystallization.  $HPO_4^{2-}$  is a weak acid and at pH levels below pH 7 the formation of  $HPO_4^{2-}$ and a low concentration of  $PO_4^{3-}$  will reduce struvite crystallization (Nelson et al., 2003). In most slurries with a typical pH of 7.5–8.3, the NH<sub>4</sub><sup>+</sup> concentration is higher than the 1:1:1 ratio ( $[Mg^{2+}]:[NH_4^+]:[PO_4^{3-}]$ ) for the formation of struvite and will not be the limiting factor in the reaction (Nelson et al., 2003). At high pH, the NH<sub>4</sub><sup>+</sup> concentration will decrease, and low NH<sub>4</sub><sup>+</sup> concentration may limit struvite formation (Buchanan et al., 1994). Optimal conditions for the sedimentation of struvite therefore occur at about pH 9 (Buchanan et al., 1994; Nelson et al., 2003).

#### 6.2.2 Technical Description

In a slurry container, the slurry may be manipulated with the purpose of forming struvite that will precipitate. Afterwards the slurry can be transferred to a thickener or another type of solid–liquid separator (see Sect. 5). In the pilot plant described by Suzuki et al. (2002), the struvite was formed in a thickener, which was also used to remove the struvite. Alternatively, the struvite may be produced in a psychrophilic anaerobic sequencing batch reactor biogas plant, where the P settles as struvite and is removed with the sludge (Massé et al., 2007a), or struvite may settle in anaerobic slurry lagoons and be removed with the sludge (Nelson et al., 2003).

#### 6.2.3 Operation and Separation Efficiency

P removal can be increased by adding Mg to the slurry being treated (Table 10). The addition of Fe and a base to the slurry will enhance dissolution of Mg and thereby increase P removal (Suzuki et al., 2002; Laridi et al., 2005). On the other hand, P removal may be low even after the addition of Mg and NH<sub>3</sub> if pH is low (Table 10; Sheffield et al., 2005).

Aeration of slurry or anaerobic digestion of slurry will increase pH and also reduce the organic matter content in the slurry (Suzuki et al., 2002; Massé et al., 2005). Both processes will greatly enhance crystallization of struvite due to an increase in the concentration of dissolved Mg<sup>2+</sup> and PO<sub>4</sub><sup>3-</sup>. Thus, aeration may produce a slurry with a mole ratio optimal for struvite crystallization, and increase struvite crystallization by a factor of about 10. In a continuous-flow pilot-scale sedimentation plant, this leads to the removal of 65–99% of P and 15% of total N (Table 10; Suzuki et al., 2002).

Crystallization is initiated by nuclei, which may be grains of sand, or by contributing energy to the nucleation process by using rakes (Battistoni et al., 2002). In most slurries, the amount of sand is generally high and one would not anticipate any need to enhance the nucleation process.

	Slurry		Separation	index $(\%)^2$	
Addition	Origin <sup>1</sup>	Dry matter (%)	N-total	P-total	Reference
Mean			16	63	
(standard deviation)			(1)	(37)	
Mg, OH <sup></sup>	Р			85	Nelson et al., 2003
Mg, NH <sub>3</sub>	Р			60	Bowers and Westerman, 2003
Mg, increase pH	Р	18		18	Suzuki et al., 2007
Mg, increase pH	Р	18		49	Suzuki et al., 2007
Mg, increase pH	Р			96	Burns et al., 2003
Mg, increase pH	Р			98	Burns et al., 2003
Fe, OH	Р	17		98	Laridi et al., 2005
Fe, OH	Р	15		99	Laridi et al., 2005
Mg, NH <sub>3</sub>	С			8	Sheffield et al., 2005
Mg, NH <sub>3</sub>	С			19	Sheffield et al., 2005

Table 10 Separation index at struvite precipitation and sedimentation

<sup>1</sup> P = pig, C = cattle.

<sup>2</sup> Separation index is the mass of a compound in the solid fraction compared to the mass of a compound in the original raw slurry.
# 7 Post-treatment: Separation Techniques

Post-treatment separation techniques need to be employed when eco-efficient technical solutions to the processing of animal slurry are sought. The objective is to produce clean water, heat and power, fuel, and highvalue N, P and K fertilizers from animal slurry – products that may not be obtained when using solid–liquid separation on its own. Some technologies have been tested on the laboratory scale, while a few have reached the farm pilot-production scale. The techniques described in this section have been chosen based on the criterion that sufficient knowledge is currently available to discuss their possible usefulness.

# 7.1 Evaporation

# 7.1.1 Technical Description, Operation and Separation Efficiency

Water and volatiles can be removed from the slurry or the liquid fraction by evaporation. The liquid is heated to boiling point, which for slurry will be a little over 100 °C at atmospheric pressure. At this temperature, both water and volatile organic compounds, such as free fatty acids and NH<sub>3</sub>, will evaporate. This vapor phase has to be condensed in order to retrieve the energy used to evaporate the water and volatiles. The energy consumed in the process is high, as the heat needed for evaporation represents roughly 670 kWh per metric ton of water evaporated. The energy (q) consumed in a single-step evaporator can thus be estimated as:

$$q = m_{water} \cdot \Delta H^{vap} \tag{18}$$

where  $\Delta H^{vap}$  is the heat of evaporation (J kg<sup>-1</sup>),  $\stackrel{\bullet}{m_{water}}$  is the amount of water removed (kg s<sup>-1</sup>), and *q* is the energy consumption (W).

To reduce the energy costs, evaporators can be operated either as single evaporators using recompressed steam, or in series where the vapor generated in the first evaporator is used as heating steam for the next evaporator. In order to transfer energy from the steam to the liquid in each evaporator, the steam temperature has to be higher than the liquid temperature. Therefore, each consecutive evaporator is operated at a lower temperature than the preceding evaporator. Boiling can thus only be achieved in each evaporator by operating each at a slightly lower pressure than the preceding evaporator. As the boiling point for slurry increases with solid content, the liquid slurry should enter the last evaporator and run counter-current to the vapor. Even though multi-step evaporation increases investment costs, experience from industrial applications shows that between three and six evaporators in series is economically viable (Foust et al., 1980; McCabe et al., 2005). In this way, a 92%-volume reduction can be achieved at an energy consumption of 120– 130 kWh per ton slurry treated (Pedersen, 2004). No literature on the effect on N separation has been found.

# 7.2 Membranes

For small particles, membrane filtration of the liquid fraction could be an attractive supplement to solid–liquid separation (see Sect. 5). Furthermore, membrane separation may be used to separate and concentrate dissolved K, P and N nutrients, producing a nutrient-rich liquid phase and, in principle, pure water for reuse or safe discharge to the environment. A review on membrane processes for slurry treatment has recently been published (Masse et al., 2007b), which gives details of efficiency and usability, and therefore only a short description is included here.

## 7.2.1 Microfiltration and Ultrafiltration

#### 7.2.1.1 Technical Description

Microfiltration removes solid particles in the range of about  $0.1-10 \,\mu$ m, while ultrafiltration retains solid particles in the range from about 5 to 200 nm (Baker, 2004). Thus, they are well suited to remove nutrients associated with small particles such as *P* (Masse et al., 2007b).

The design of membrane units is based on the membrane flux ( $J_{liquid}$ ), which is the amount of liquid removed per area of membrane. In micro- and ultrafiltration,  $J_{liquid}$  can be described by the general equation:

$$J_{liquid} = \frac{\Delta P}{\mu_{liquid} \cdot (R_m + R_{rev} + R_{irrev})}$$
(19)

where *J* is the flux (kg m<sup>-2</sup> s<sup>-1</sup>),  $\mu_{liquid}$  is the viscosity of the permeate,  $R_m$  is the resistance to flow from the membrane (m<sup>-1</sup>),  $R_{rev}$  is the reversible fouling resistance often ascribed to filter cake formation and concentration polarization (m<sup>-1</sup>),  $R_{irrev}$  is the irreversible fouling resistance normally ascribed to pore blocking and adsorption of materials to the membrane (m<sup>-1</sup>), and  $\Delta P$  is the transmembrane pressure over the membrane (Pa).

During micro- and ultrafiltration, the slurry is pressed through a porous membrane. If dead-end filtration is used, a very dense filter cake build-up will occur. Therefore, micro- and ultrafiltration have to be carried out as cross-flow filtration, where only a fraction of the liquid is removed as permeate through the membrane, while solids and part of the liquid are retained as retentate. The cross flow removes most of the solids deposited on the surface, but some flow-controlled reversible filter cake formation cannot be avoided. Worse still, irreversible adhesion of minor particles can occur in the membrane pores, partly blocking the path for the liquid. In addition, bacterial growth can occur on the membrane surface, further reducing the flow through the membrane. This kind of membrane fouling will be especially severe for the treatment of slurry. Reversible filter cake formation can be removed by flushing with water. In contrast, the irreversible fouling can, at best, only partly be removed by cleaning the membrane in the system (CIP) with dilute base followed by dilute acid and, if necessary, enzymatic treatment.

Because of fouling, micro- and ultrafiltration membranes can only be used to separate pre-treated slurry, such as effluent streams from biogas reactors or runoff streams from centrifuges.

#### 7.2.1.2 Operation and Separation Efficiency

For slurry microfiltration, transmembrane pressure will typically be around 100–180 kPa. At this pressure, filter cake formation and fouling become rate-determining, and increasing the transmembrane pressure no longer increases the flux over the membrane to any significant extent. Increasing the feed flow can increase the flux by reducing the filter cake thickness, but at velocities above around 2 m s<sup>-1</sup> this becomes uneconomic (Owen et al., 1995). Therefore, the maximum flux attainable is limited to around

160  $\text{L}\cdot\text{m}^{-2}\cdot\text{h}^{-1}$ , although long-term fluxes down to 10–40  $\text{L}\cdot\text{m}^{-2}\cdot\text{h}^{-1}$  should be expected, due to fouling. Depending on the size distribution, a retention efficiency of 75% of *DM* can be expected (Masse et al., 2007b). Membrane filtration retains small particles; therefore, the large amount of P contained in particles between 0.45 and 10  $\mu$ m is retained with membrane filtration, and a high P removal can be expected (Masse et al., 2007b). Dissolved N, K and *DM* will not be retained.

For ultrafiltration, transmembrane pressures up to 800 kPa can be used, depending on membrane pore size. The larger the pore size, the lower the transmembrane pressure (Baker, 2004). For filtered pig slurry, fluxes between 10 and 40  $L \cdot m^{-2} \cdot h^{-1}$  at a transmembrane pressure of 100 kPa have been reported with 100% *DM* removal and a P removal efficiency of up to 87%, depending on operational conditions (Masse et al., 2007b). For partially digested non-filtered pig slurry, flushing the membrane every 5 minutes with permeate leads to a fall in flux of 25% over a 70-day period compared with a flux decline of 75% if only a single cleaning procedure is carried out (du Preez et al., 2005). In this study, though, the maximum flux was only 16  $L \cdot m^{-2} \cdot h^{-1}$ .

## 7.2.2 Nanofiltration and Reverse Osmosis

#### 7.2.2.1 Technical Description

The liquid fraction from an ultrafiltration membrane can contain large amounts of dissolved K<sup>+</sup>, NH<sub>4</sub><sup>+</sup> and NH<sub>3</sub>. Reverse osmosis and, to some extent, nanofiltration membranes can retain dissolved nutrients, and deliver high concentrate retentate and purified water permeate. Depending on the membrane chosen, nanofiltration will retain uncharged molecules larger than 200–400 Da and, to a lesser extent, smaller charged molecules such as Ca<sup>2+</sup>, Mg<sup>2+</sup> and NH<sub>4</sub><sup>+</sup>. Nanofiltration can thus be used to remove *DM* and, to some extent, charged molecules. If a pure water permeate is needed, reverse osmosis can retain NH<sub>4</sub><sup>+</sup> and K<sup>+</sup> and, to a lesser extent, NH<sub>3</sub> (Masse et al., 2007b).

The fouling problems are even more severe for nanofiltration and reverse osmosis than for micro- and ultrafiltration. In connection with slurry, nanofiltration or reverse osmosis can therefore only be used for separation of dissolved components from the permeate produced in an ultrafiltration unit. The direct filtration of raw slurry will foul the membrane within minutes.

The water flux through nanofiltration and reverse osmosis membranes can be described (Baker, 2004) as follows:

$$J_{liquid} = A \cdot (\Delta p - \Delta \Pi) \tag{20}$$

where A is the water permeability (kg m<sup>-2</sup> s<sup>-1</sup>Pa<sup>-1</sup>) and  $\Delta\Pi$  (Pa) is the osmotic pressure difference between retentate and permeate. The maximum possible retentate concentration will be determined by the osmotic pressure difference and the maximum possible transmembrane pressure.

As concentration polarization influences the osmotic pressure on the membrane surface during nanofiltration and reverse osmosis, experimental permeability constants depend on the liquid cross flow and, because of unavoidable fouling, on time in use. As for micro- and ultrafiltration membranes, cleaning will need to be performed regularly in order to avoid fouling.

## 7.2.2.2 Operation and Separation Efficiency

For nanofiltration transmembrane operation, pressure will be around 350–3000 kPa. Although not well suited for the removal of K or N, nanofiltration membranes have been shown to be capable of removing up to 52% of the  $NH_4^+$  and 78% of the K (Masse et al., 2007b) and, if a suitable membrane is chosen, all the soluble *DM* with a molecular weight above 200 Da will be removed.

For more demanding applications, reverse osmosis has to be used. The transmembrane pressure for reverse osmosis operations is typically around 3.5-6.5 MPa, although up to 150 MPa can be achieved for specially designed membranes. The retention of K is usually high and independent of pH but, as NH<sub>4</sub><sup>+</sup> is retained better than NH<sub>3</sub>, the separation is very pH-dependent and also depends on the ionic strength of the retentate (Masse et al., 2008). Therefore, N separation will depend on pH, slurry origin and the final volume reduction of the retentate. For pig slurry, the retention of NH<sub>4</sub><sup>+</sup> thus decreased from 90% to 70% and for K from 93% to 87% when the reduction in retentate volume was increased from 50% to 90% (Masse et al., 2007b). As the flux for nanofiltration and reverse osmosis is very dependent not only on fouling but also on the osmotic pressure of the retentate, the flux will change dramatically during a concentration process, with maximum fluxes reaching up to 65 L m<sup>-2</sup> h<sup>-1</sup> at the initial conditions, but approaching zero at the final reduction volume, where the osmotic pressure approaches the transmembrane pressure.

# 8 Conclusions

Separation indexes, which express the distribution of the specific component between the solid and liquid fractions (see Eqs. (5, 6)), have proved useful for the comparison of the efficiency of the different separation techniques. From a literature review it was found that only about 25% of the papers on solid–liquid separation included sufficient data to assess the separation index of the technology tested. This emphasizes the need for a standard method for characterizing slurry separation efficiency, including a thorough characterization of the animal slurry treated and the fractions produced, and flow measurements of the slurry being treated.

Slurry can be treated with separation technologies that produce a solid fraction in which much of the P and dry matter (DM) and significant amounts of the N in the slurry are retained. These technologies are simple to use and reliable. More complex technologies may, although with a relatively high energy consumption, remove nitrogen, including NH<sub>4</sub><sup>+</sup>, from the slurry. By combining solid–liquid separation technologies with pre- and post-treatments, end-products having an optimal composition for a specific end-use may be produced.

In order to achieve these objectives, there is a need for studies on the physical and chemical properties of slurry and their effects on separation efficiency, which include a functional physical-chemical characterization of slurry and identification of how the treatment affects the physical-chemical characteristics of the separation product. Only if this knowledge is available can existing separation technologies be optimized and combined in a systematic way in order to achieve the desired end-products.

At present, the most efficient solid–liquid mechanical separators for the removal of DM, P and, to some extent, total N and  $NH_4^+$ , can be ranked in the following order: centrifuge > sedimentation > filtration without pressure > filtration with pressure. Struvite crystallization prior to the solid–liquid separation improves the N and P separation. Flocculation before separation also improves the separation of DM, N and P significantly. The best flocculant available at present (polyacrylamide – PAM) is a cationic, medium-charge-density (20–40 mol%), linear, large-molecular-weight polymer, and the best coagulants are Al<sub>2</sub>(SO<sub>4</sub>)<sub>3</sub> and FeCl<sub>3</sub>. Filtration without pressure is the best slurry separation technique for slurry with added flocculants.

After the separation of DM, P and N from the slurry, the liquid fraction produced may be treated with membrane separation, evaporation of water or NH<sub>3</sub> stripping. Regarding membrane separation, attention must be paid to fouling problems. Stripping and evaporation have a high energy consumption, and efficient heat recovery or cheap surplus heat is necessary for these technologies to be of interest at present.

In order to set up an animal slurry separation operation, a whole-system approach is important either with or without pre- and post-treatments. Just as in any industrial farm operation, it is of paramount importance to balance investment, running and labor costs against product added value, environmental impact and process complexity. This seems to be a promising area for future intensive research and development.

Acknowledgements This study was supported by a grant from the Danish Agency for Science, Technology and Innovation to the 'CLEANWASTE' project, under the Environmentally Friendly Technologies program.

# References

- Amon B., Kryvoruchko V., Amon T., Zechmeister-Boltenstern S. (2006) Methane, nitrous oxide and ammonia emissions during storage and after application of dairy cattle slurry and influence of slurry treatment, Agr. Ecosyst. Environ. 112, 153–162.
- Angelidaki I., Ellegaard L., Ahring B.K. (1993) A mathematical model for dynamic simulation of anaerobic digestion of complex substrates: focusing on ammonia inhibition, Biotechnol. Bioeng. 42, 159–166.
- Arogo J., Westerman P.W., Heber A.J. (2003) A review of ammonia emissions from confined swine feeding operations, Trans. ASAE 46, 805–817.
- Ashmore M., Hearn J. (2000) Flocculation of model latex particles by chitosans of varying degrees of acetylation, Langmuir 16, 4906–4911.

- Baker R.W. (2004) Membrane Technology and Applications, 2nd ed., John Wiley & Sons, Chichester, UK.
- Barrow J.T., Van Horn H.H., Anderson D.L., Nordstedt R.A. (1997) Effects of Fe and Ca additions to dairy wastewaters on solids and nutrient removal by sedimentation, Appl. Eng. Agric. 13, 259–267.
- Battistoni P., De Angelis A., Prisciandaro M., Boccadoro R., Bolzonella D. (2002) P removal from anaerobic supernatants by struvite crystallization: long term validation and process modelling, Water Res. 36, 1927–1938.
- Bernal M.P., Roig A., Madrid R., Navarro A.F. (1992) Salinity risks on calcareous soils following pig slurry applications, Soil Use Manage. 8, 125–130.
- Bernal M.P., Roig A., García D. (1993) Nutrient balances in calcareous soils after application of different rates of pig slurry, Soil Use Manage. 9, 9–14.
- Bouyer F., Robben A., Yu W.L., Borkovec M. (2001) Aggregation of colloidal particles in the presence of oppositely charged polyelectrolytes: effect of surface charge heterogeneities, Langmuir 17, 5225–5231.
- Bouwman A.F., Booij H. (1998) Global use and trade of foodstuffs and consequences for the nitrogen cycle, Nutr. Cycl. Agroecosys. 52, 261–267.
- Bowers K.E., Westerman P.W. (2003) Phosphorus removal in novel fluidized bed, Proceedings of the 2003 Annual Meeting of the American Society of Agricultural Engineers, Paper Number 03–4123, ASAE, St. Joseph, MI.
- Bratskaya S., Schwarz S., Laube J., Liebert T., Heinze T., Krentz O., Lohmann C., Kulicke W.M. (2005) Effect of polyelectrolyte structural features on flocculation behavior: cationic polysaccharides vs. synthetic polycations, Macromol. Mater. Eng. 290, 778–785.
- Bril J., Salomons W. (1990) Chemical composition of animal manure: a modelling approach, Neth. J. Agr. Sci. 38, 333–351.
- Buchanan J.R., Mote C.R., Robinson R.B. (1994) Struvite control by chemical treatment, Trans. ASAE 37, 1301–1308.
- Burns R.T., Moody L.B., Celen I., Buchanan J.R. (2003) Optimization of phosphorus precipitation from swine manure slurries to enhance recovery, Water Sci. Technol. 48, 139–146.
- Burton C.H., Turner C. (2003) Manure Management: Treatment Strategies for Sustainable Agriculture, 2nd ed., Silsoe Research Institute, UK.
- Canh T.T., Aarnink A.J.A., Mroz Z., Jongbloed A.W., Schrama J.W., Verstegen M.W.A. (1998) Influence of electrolyte balance and acidifying calcium salts in the diet of growingfinishing pigs on urinary pH, slurry pH and ammonia volatilisation from slurry, Livest. Prod. Sci. 56, 1–13.
- Chadwick D.R., John F., Pain B.F., Chambers B.J., Williams J. (2000) Plant uptake of nitrogen from the organic nitrogen fraction of animal manures: a laboratory experiment, J. Agr. Sci. 134, 159–168.
- Chapuis-Lardy L., Fiorini J., Toth J., Dou Z. (2004) Phosphorus concentration and solubility in dairy feces: variability and affecting factors, J. Dairy Sci. 87, 4334–4341.
- Christensen G.L., Dick R.I. (1985) Specific resistance measurements: nonparabolic data, J. Environ. Eng. ASCE 111, 243–257.

- Christensen M.L., Keiding K. (2007) Filtration model for suspensions that form filter cakes with creep behaviour, AIChE J 53, 598–609.
- Christensen M.L., Hjorth M., Keiding K. (2009) Characterization of pig slurry with reference to flocculation and separation, Water Res. 43, 773–783.
- Clemens J., Trimborn M., Weiland P., Amon B. (2006) Mitigation of greenhouse gas emissions by anaerobic digestion of cattle slurry, Agr. Ecosyst. Environ. 112, 171–177.
- Conn K.L., Topp E., Lazarovits G. (2007) Factors influencing the concentration of volatile fatty acids, ammonia, and other nutrients in stored liquid pig manure, J. Environ. Qual. 36, 440–447.
- Converse J.C., Karthikeyan K.G. (2004) Nutrient and solids separation of flushed dairy manure by gravity settling, Appl. Eng. Agric. 20, 503–507.
- Dao T.H., Daniel T.C. (2002) Particulate and dissolved phosphorus chemical separation and phosphorus release from treated dairy manure, J. Environ. Qual. 81, 1388–1398.
- DeBusk J.A., Arogo Ogejo J., Knowlton K.F., Love N.G. (2008) Chemical phosphorus removal for separated flushed dairy manure, Appl. Eng. Agric. 24, 499–506.
- Dou Z., Ferguson J.D., Fiorini J., Toth J.D., Alexander S.M., Chase L.E., Ryan C.M., Knowlton K.F., Kohn R.A., Peterson A.B., Sims J.T., Wu Z. (2003) Phosphorus feeding levels and critical control points on dairy farms, J. Dairy Sci. 86, 3787–3795.
- du Preez J., Norddahl B., Christensen K. (2005) The BIOREK((R)) concept: a hybrid membrane bioreactor concept for very strong wastewater, Desalination 183, 407–415.
- Eriksson L., Alm B., Stenius P. (1993) Formation and structure of polystyrene latex aggregates obtained by flocculation with cationic polyelectrolytes. 1. Adsorption and optimum flocculation concentrations. Colloid. Surface. A 70, 47–60.
- Estevez Rodríguez M.D., Gomez del Puerto A.M., Montealegre Meléndez M.L., Adamsen A.P.S., Gullov P., Sommer S.G. (2005) Separation of phosphorus from pig slurry using chemical additives, Appl. Eng. Agric. 21, 739–742.
- FAOSTAT (2006) New Release, 27 July 2006 [available at http:// faostat.fao.org].
- Fordham A.W., Schwertmann U. (1977a) Composition and reactions of liquid manure (gülle), with particular reference to phosphate. I. Analytical composition and reaction with poorly crystalline iron oxide (ferrihydrite), J. Environ. Qual. 6, 133–136.
- Fordham A.W., Schwertmann U. (1977b) Composition and reactions of liquid manure (gülle), with particular reference to phosphate. II. Solid phase components, J. Environ. Qual. 6, 136–140.
- Fordham A.W., Schwertmann U. (1977c) Composition and reactions of liquid manure (gülle), with particular reference to phosphate. III. pH-buffering capacity and organic components, J. Environ. Qual. 6, 140–144.
- Foust A.S., Wenzel L.A., Clump C.W., Maus L., Andersen L.B. (1980) Principles of Unit Operations, 2nd ed., John Wiley & Sons, New York.
- Gao Y.C., Liao P.H., Lo K.V. (1993) Chemical treatment of swine wastewater, J. Environ. Sci. Health A28, 795–807.
- Garcia M.C., Vanotti M.B., Szogi A.A. (2007) Simultaneous separation of phosphorus sludge and manure solids with polymers, Trans. ASABE 50, 2205–2215.

- Garcia M.C., Szogi A.A., Vanotti, M.B., Chastain J.P., Millner P.D. (2009) Enhanced solid–liquid separation of dairy manure with natural flocculants, Bioresource Technol. DOI: 10.1016/j.biortech.2008.11.012.
- Gilbertson C.B., Nienaber J.A. (1978) Separation of coarse solids from beef-cattle manure, Trans. ASAE 21, 1185–1188.
- Greaves J., Hobbs P., Chadwick D., Haygarth P. (1999) Prospects for the recovery of phosphorus from animal manures: a review, Environ. Technol. 20, 697–708.
- Gregory J. (1973) Rates of flocculation of latex particles by cationic polymers, J. Colloid Interf. Sci. 42, 448–456.
- Gregory J. (1989) Fundamentals of flocculation, Crit. Rev. Env. Cont. 19, 185–229.
- Gungor K., Karthikeyan K.G. (2005) Probable phosphorus solid phases and their stability in anaerobically digested dairy manure, Trans. ASAE 48, 1509–1520.
- Han I.K., Lee J.H., Piao X.S., Li D.F. (2001) Feeding and management system to reduce environmental pollution in swine production: review, Asian-Australasian J. Anim. Sci. 14, 432–444.
- Hansen M.N., Henriksen K., Sommer S.G. (2006) Observations of production and emission of greenhouse gases and ammonia during storage of solids separated from pig slurry: effects of covering, Atmos. Environ. 40, 4172–4181.
- Hegg R.O., Larson R.E., Moore J.A. (1981) Mechanical liquidsolid separation in beef, dairy and swine waste slurries, Trans. ASAE 24, 159–163.
- Henriksen K., Berthelsen L., Matzen R. (1998a) Separation of liquid pig manure by flocculation and ion exchange. Part 1. Laboratory experiments, J. Agr. Eng. Res. 69, 115–125.
- Henriksen K., Berthelsen L., Matzen R. (1998b) Separation of liquid pig manure by flocculation and ion exchange. Part 2. Pilot-scale system, J. Agr. Eng. Res. 69, 127–131.
- Henze M., Harremoës P., la Cour Jansen J., Arvin E. (1997) Treatment plants for phosphorus removal from wastewater, in: Henze M., Harremoës P., la Cour Jansen J., Arvin E. (Eds.), Wastewater Treatment: Biological and Chemical Processes, Springer, Berlin, pp. 311–351.
- Hindrichsen I.K., Wettstein H.-R., Machmüller A., Kreuzer M. (2006) Methane emission, nutrient degradation and nitrogen turnover in dairy cows and their slurry at different milk production scenarios with and without concentrate supplementation, Agr. Ecol. Environ. 113, 150–161.
- Hjorth M., Christensen M.L. (2008) Evaluation of methods to determine the flocculation optimum at manure separation, Trans. ASABE 51, 2093–2103.
- Hjorth M., Christensen M.L., Christensen P.V. (2008) Flocculation, coagulation and precipitation of manure affecting three separation techniques, Bioresource Technol. 99, 8598–8604.
- Hjorth M., Nielsen A.M., Nyord T., Hansen M.N., Nissen P., Sommer S.G. (2009) Nutrient value, odour emission and energy production of manure as influenced by anaerobic digestion and separation, Agron. Sustain. Dev. 29, 329–338.
- Holmberg R.D., Hill D.T., Prince T.J., Van Dyke, N.J. (1983) Potential of solid-liquid separation of swine wastes for methane production, Trans. ASAE 26, 1803–1807.
- Husted S., Jensen L.S., Storgaard Jørgensen S. (1991) Reducing ammonia loss from cattle slurry by the use of acidifying additives: the role of the buffer system, J. Sci. Food Agr. 57, 335–349.

- Japenga J., Harmsen K. (1990) Determination of mass balances and ionic balances in animal manure, Neth. J. Agr. Sci. 38, 353–367.
- Karr P.R., Keinath T.M (1978) Influence of particle size on sludge dewaterability, J. WPCF 50, 1911–1930.
- Karthikeyan K.G., Kalbasi M., Miller P.S. (2005) Nitrogen and solution dynamics in soils receiving chemically treated dairy manure, Trans. ASAE 48, 601–610.
- Kayser R. (1997) Technologische bewertung der FuE: Vorhaben in: KTBL (Eds.), Statusseminar "Umweltverträgliche Gülleaufbereitung und -verwertung": Congress Proceedings, Arbeitspapier Nr. 242. Münster Hiltrup, Germany, pp. 24–30.
- Kirchmann H., Witter E. (1989) Ammonia volatilization during aerobic and anaerobic manure decomposition, Plant Soil 115, 35–41.
- Krieter J. (2002) Evaluation of different pig production systems including economic, welfare and environmental aspects, Arch. Tierz. 45, 223–235.
- Kroodsma W. (1986) Separation and removal of faeces and urine using filter nets under slatted floors in piggeries, J. Agr. Eng. Res. 34, 75–84.
- Krumpelman B.W., Daniel T.C., Edwards F.G., McNew R.W., Miller D.M. (2005) Optimum coagulant and flocculant concentrations for solids and phosphorus removal from prescreened flushed dairy manure, Appl. Eng. Agric. 21, 127–135.
- Lachance I., Godbout S., Marquis A., Hamel D., Larouche J.-P. (2004) Would separating feces and urine under pigs be a key to minimize odors and gas emissions? ASAE/CSAE Meeting Paper No. 0444160, Fairmont Chateau Laurier, Ottawa, ON, Canada, p. 17.
- Landry H., Laguë C., Roberge M. (2004) Physical and rheological properties of manure products, Appl. Eng. Agric. 20, 277–288.
- Laridi R., Auclair J.C., Benmoussa H. (2005) Laboratory and pilot-scale phosphate and ammonium removal by controlled struvite precipitation following coagulation and flocculation of swine wastewater, Environ. Technol. 26, 525–536.
- Larsson A., Walldal C., Wall S. (1999) Flocculation of cationic polymers and nanosized particles, Colloid. Surface. A 159, 65–76.
- Le H.-C. (1998) Biodigester effluent versus manure, from pigs or cattle, as fertilizer for duckweed (*Lemna* spp.), Livest. Res. Rural Dev. 10 [available at http://www.fao.org/AG/aGa/ agap/FRG/FEEDback/Irrd/Irrd10/3/chau2.htm].
- Loughrin J.H., Szogi A.A., Vanotti M.B. (2006) Reduction of malodorous compounds from liquid swine manure by a multi-staged treatment system, Appl. Eng. Agric. 22, 867–873.
- Lyklema J. (1977) Water at interfaces: a colloid chemical approach, J. Colloid Interf. Sci. 58, 242.
- Massé L., Massé D.I., Beaudette V., Muir M. (2005) Size distribution and composition of particles in raw and anaerobically digested swine manure, Trans. ASAE 48, 1943–1949.
- Massé D.I., Croteau F., Massé L. (2007a) The fate of crop nutrients during digestion of swine manure in psychrophilic anaerobic sequencing batch reactors, Bioresource Technol. 98, 2819–2823.

- Masse L., Massé D.I., Pellerin Y. (2007b) The use of membranes for the treatment of manure: a critical literature review, Biosyst. Eng. 98, 371–380.
- Masse L., Massé D.I., Pellerin Y. (2008) The effect of pH on the retention of swine manure nutrients by reverse osmosis membranes, 10th World Filtration Congress II, pp. 276–280.
- McCabe W.L., Smith J.C., Harriott P. (2005) Unit Operations of Chemical Engineering, 7th ed., McGraw-Hill, New York.
- Menear J.R., Smith L.W. (1973) Dairy-cattle manure liquid-solid separation with a screw press, J. Anim. Sci. 36, 788–791.
- Menzi H. (2002) Manure management in Europe: results of a recent survey, in: Proceedings of the 10th Conference of the FAO/ESCORENA Network on Recycling Agricultural, Municipal and Industrial Residues in Agriculture (RAMIRAN), Strbske Pleso, Slovak Republic, 14–18 May, pp. 93–102.
- Meyer D., Ristow P.L., Lie M. (2007) Particle size and nutrient distribution in fresh dairy manure, Appl. Eng. Agric. 23, 113–117.
- Mikkelsen L.H. (1999) A physico-chemical approach to the floc strength concept: with dewatering implications, PhD thesis, Environmental Engineering Laboratory, Aalborg University, Denmark.
- Moore J.A., Hegg R.O., Scholz D.C., Strauman E. (1975) Settling solids in animal waste slurries, Trans. ASAE 18, 694–698.
- Møller H.B., Lund I., Sommer S.G. (2000) Solid–liquid separation of livestock slurry: efficiency and cost, Bioresource Technol. 74, 223–229.
- Møller H.B., Sommer S.G., Ahring B.K. (2002) Separation efficiency and particle size composition in relation to manure type and storage conditions, Bioresource Technol. 85, 189–196.
- Møller H.B., Ahring B.K., Sommer S.G. (2004) Methane productivity of manure, straw and solid fractions of manure, Biomass Bioenerg. 26, 485–495.
- Møller H.B., Hansen J.D., Sørensen C.A.G. (2007a) Nutrient recovery by solid-liquid separation and methane productivity of solids, Trans. ASABE 50, 193–200.
- Møller H.B., Jensen H.S., Tobiasen L., Hansen M.N. (2007b) Heavy metal and phosphorus content of fractions from manure treatment and incineration, Environ. Technol. 28, 1403–1418.
- Nahm K.H (2005) Environmental effects of chemical additives used in poultry litter and swine manure, Crit. Rev. Environ. Sci. Technol. 35, 487–513.
- Ndegwa P.M., Zhu J., Luo A.C. (2001) Effects of solid levels and chemical additives on removal of solids and phosphorus in swine manure, J. Environ. Eng. 127, 1111–1115.
- Nelson N.O., Mikkelsen R.L., Hesterberg D.L. (2003) Struvite precipitation in anaerobic swine lagoon liquid: effect of pH and Mg:P ratio and determination of rate constant, Bioresource Technol. 89, 229–236.
- Nowostawska U., Sander S.G., McGrath K.M., Hunter K.A. (2005) Effect of trivalent metal sulfates on the coagulation and particle interactions of alumina colloids, Colloid. Surface. A 266, 200–206.
- Nørgaard P. (2006) Partikellængde og strukturværdi i majsensilage, in: Sehested J. (Ed.), Temamøde om kvægernæring – Kvalitet af majsensilage – Aktuel forskning. Intern, Rapport Husdyrbrug 46, 19–27, Danish Institute of Agricultural Sciences.

- Oenema O., Bannink A., Sommer S.G., Velthof G. (2001) Gaseous nitrogen emission from livestock farming system, in: Follett R.F., Hatfield J.L. (Eds.), Nitrogen in the Environment: Sources, Problems and Management, Elsevier Science B.V., Amsterdam, pp. 255–289.
- Ohlinger K.N., Young T.M., Schroeder E.D. (2000) Postdigestion struvite precipitation using a fluidized bed reactor, J. Environ. Eng. 126, 361–368.
- Ong H.K. (2002) Livestock waste management in Southeast Asia, in: Ong H.K., Zulkifli I., Tee T.P., Liang J.B. (Eds.), Global Perspective in Livestock Waste Management: Proceedings of the 4th International Livestock Waste Management Symposium and Technology Expo, Penang, Malaysia, pp. 59–67.
- Otto E.R., Yokoyama M., Hengemuehle S., von Bermuth R.D, van Kempen T., Trottier N.L. (2003) Ammonia, volatile fatty acids, phenolics, and odor offensiveness in manure from growing pigs fed diets reduced in protein concentration, J. Anim. Sci. 81, 1754–1763.
- Owen G., Bandi M., Howell J.A., Churchouse S.J. (1995) Economic assessment of membrane processes for water and waste water treatment, J. Membrane Sci. 102, 77–91.
- Pain B.F., Hepherd R.Q., Pitman R.J. (1978) Factors affecting the performances of four slurry separating machines, J. Agr. Eng. Res. 23, 231–242.
- Pedersen K.F. (2004) Strategies for nitrogen removal from liquid manure, Master's thesis, Institute of Chemistry and Physics, University of Southern Denmark.
- Petersen J., Kjellerup V. (1990) Fordeling af næringsstoffer i gyllebeholdere. Statens Planteavlsforsøg, Grøn Viden, Landbrug Nr. 47.
- Petersen S.O., Sommer S.G., Béline F., Burton C., Dach J., Dourmad J.Y., Leip A., Misselbrook T., Nicholson F., Poulsen H.D., Provolo G., Sørensen P., Vinnerås B., Weiske A., Bernal M.-P., Böhm R., Juhász C., Mihelic R. (2007) Recycling of livestock wastes and other residues: a whole-farm perspective, Livest. Sci. 112, 180–191.
- Pfeiffer A. (1995) The influence of phytase in phosphorusreduced diets on the amount of P-excretion in the case of fattening pigs and estimations of the amount of P-excretion based on P-concentration in feces respectively, Z. Zuchtungsk. 67, 147–157.
- Pieters J.G., Neukermans G.G.J., Colanbeen M.B.A. (1999) Farm-scale membrane filtration of sow slurry, J. Agr. Eng. Res. 73, 403–409.
- Poels J., Vanrompu K., Verstraete W. (1988) Concentration of liquid piggery manure by membrane separation techniques, Rev. Agr. 41, 935–951.
- Portejoie S., Dourmad J.Y., Martinez J., Lebreton Y. (2004) Effect of lowering dietary crude protein on nitrogen excretion, manure composition and ammonia emission from fattening pigs, Livest. Prod. Sci. 91, 45–55.
- Pos J., Trapp R., Harvey M. (1984) Performance of a brushed screen roller press manure separator, Trans. ASAE 27, 1112–1118.
- Poulsen H.D. (2000) Phosphorus utilization and excretion in pig production, J. Environ. Qual. 29, 24–27.
- Poulsen H.D., Børsting C.F., Rom H.B., Sommer S.G. (2001) Nitrogen, Phosphorus and Potassium in Animal Manure: Norm Values 2000, DIAS Report: Animal Manure No. 36,

Ministry of Food, Agriculture and Fisheries, Danish Institute of Agricultural Sciences, Tjele, Denmark [in Danish].

- Powers W.J., Flatow L.A. (2002) Flocculation of swine manure: influence of flocculant, rate of addition, and diet, Appl. Eng. Agric. 18, 609–614.
- Rattanarajcharkul R., Rucha W., Sommer S.G., Menzi H. (2001) Area-wide integration of specialised livestock and crop production in eastern Thailand, in: Sangiorgi F. (Ed.), Technology Transfer: Proceedings of the 9th International Conference of the FAO ESCORENA Network on Recycling of Agricultural, Municipal and Industrial Residues in Agriculture (RAMIRAN), 6–9 September, Gargano, Italy.
- Reimann W. (1989) Fest-flüssig-trennung von gülle und gülleaufbereitungsprodukten, Arch. Acker- Pflanzenbau Bodenkd. (Berlin) 33, 617–625.
- Rico J.L., Garcia H., Rico C., Tejero I. (2007) Characterisation of solid and liquid fractions of dairy manure with regard to their component distribution and methane production, Bioresource Technol. 98, 971–979.
- Rubæk G.H., Henriksen K., Petersen J., Rasmussen B., Sommer S.G. (1996) Effects of application technique and anaerobic digestion on gaseous nitrogen loss from animal slurry applied to ryegrass (*Lolium perenne*), J. Agr. Sci. 126, 481–492.
- Saeys W., Mouazen A.M., Ramon H. (2005) Potential for onsite and online analysis of pig manure using visible and near infrared reflectance spectroscopy, Biosyst. Eng. 91, 393–402.
- Schechter L.-A., Deskin R., Essenfeld A., Bernard B., Friedman M., Grube E. (1995) Evaluation of the toxicological risk associated with the use of polyacrylamides in the recovery of nutrients from food processing waste (II), Int. J. Toxicol. 14, 34–39.
- Severin B.F., Grethlein H.E. (1996) Laboratory simulation of belt press dewatering: application of the Darcy equation to gravity drainage, Water Environ. Res. 68, 359–369.
- Sheffield R., Bowers K., Harrison J. (2005) Phosphorus removal on dairies in the Pacific Northwest: applying a cone shaped fluidized bed phosphorus crystallizer, in: The Development of Alternative Technologies for the Processing and Use of Animal Waste: Proceedings of the Animal Waste Management Symposium, Research Triangle Park, NC [available at http://www.cals.ncsu.edu/waste\_mgt/ natlcenter/sanantonio/Sheffield.pdf].
- Sherman J.J., Van Horn H.H., Nordstedt R.A. (2000) Use of flocculants in dairy wastewaters to remove phosphorus, Appl. Eng. Agric. 16, 445–452.
- Smith-Palmer T., Campbell N., Bowman J.L., Dewar P. (1994) Flocculation behavior of some cationic polyelectrolytes, J. Appl. Polym. Sci. 52, 1317–1325.
- Sneath R.W., Shaw M., Williams A.G. (1988) Centrifugation for separating piggery slurry. 1. The performance of a decanting centrifuge, J. Agr. Eng. Res. 39, 181–190.
- Sommer S.G., Husted S. (1995a) The chemical buffer system in raw and digested animal slurry, J. Agr. Sci. 124, 45–53.
- Sommer S.G., Husted S. (1995b) A simple model of pH in slurry, J. Agr. Sci. 124, 447–453.
- Sommer S.G., Hutchings N.J. (2001) Ammonia emission from field-applied manure and its reduction: invited paper, Eur. J. Agron. 15, 1–15.

- Sommer S.G., Mathanpal G., Dass T.T. (2005) A simple biofilter for treatment of pig slurry in Malaysia, Environ. Technol. 26, 303–312.
- Sommer S.G., Zhang G.Q., Bannink A., Chadwick D., Hutchings N.J., Misselbrook T., Menzi H., Ni J.-Q., Oenema O., Webb J., Monteny G.-J. (2006) Algorithms determining ammonia emission from livestock houses and manure stores, Adv. Agron. 89, 261–335.
- Sommer S.G., Petersen S.O., Sørensen P., Poulsen H.D., Møller H.B. (2007) Greenhouse gas emission and nitrogen turnover in stored liquid manure, Nutr. Cycl. Agroecosys. 78, 27–36.
- Sommer S.G., Maahn M., Poulsen H.D, Hjorth M., Sehested J. (2008) Interactions between phosphorus feeding strategies for pig and dairy cows and separation efficiency of slurry, Environ. Technol. 29, 75–80.
- Sørensen P. (1998) Effects of storage time and straw content of cattle slurry on the mineralization of nitrogen and carbon in soil, Biol. Fert. Soils 27, 85–91.
- Sørensen P.B., Christensen J.R., Bruus J.H. (1995) Effect of small scale solids migration in filter cakes during filtration of wastewater solids suspensions, Water Environ. Res, 67, 25–32.
- Sørensen C.G., Jacobsen B.H., Sommer S.G. (2003) An assessment tool applied to manure management systems using innovative technologies, Biosyst. Eng. 86, 315–325.
- Sørensen C.G., Møller H.B. (2006) Operational and economic modelling and optimization of mobile slurry separation, Appl. Eng. Agric. 22, 185–193.
- Steinfeld H., Gerber P., Wassenaar T., Castel V., Rosales M., de Haan C. (2006) Livestock's Long Shadow: Environmental Issues and Options, FAO, Rome.
- Suzuki K., Tanaka Y., Osada T., Waki M. (2002) Removal of phosphate, magnesium, and calcium from swine wastewater through crystallization enhanced by aeration, Water Res. 36, 2991–2998.
- Suzuki K., Tanaka Y., Kuroda K., Hanajima D., Fukumoto Y., Yasuda T., Waki M. (2007) Removal and recovery of phosphorous from swine wastewater by demonstration crystallization reactor and struvite accumulation device, Bioresource Technol. 98, 1573–1578.
- Szögi A.A., Vanotti M.B., Hunt P.G. (2006) Dewatering of phosphorus extracted from liquid swine waste, Bioresource Technol. 97, 183–190.
- Taiganides E.P. (1992) Pig Waste Management and Recycling: The Singapore Experience, International Development Research Centre, Ottawa, ON, Canada.
- Thomsen I.K. (2000) C and N transformation in <sup>15</sup>N cross-labelled ruminant manure during anaerobic and aerobic storage, Bioresource Technol. 72, 267–274.

- Timby G.G., Daniel T.C., McNew R.W., Moore P.A. (2004) Polymer type and aluminum chloride affect screened solids and phosphorus removal from liquid dairy manure, Appl. Eng. Agric. 20, 57–64.
- Vadas P.A. (2006) Distribution of phosphorus in manure slurry and its infiltration after application to soils, J. Environ. Qual. 35, 542–547.
- Vanotti M.B., Hunt P.G. (1999) Solids and nutrient removal from flushed swine manure using polyacrylamides, Trans. ASAE 42, 1833–1840.
- Vanotti M.B., Rashash D.M.C., Hunt P.G. (2002) Solid-liquid separation of flushed swine manure with PAM: Effect of wastewater strenght, Trans. ASAE 45, 1959–69.
- Vanotti M.B., Rice J.M., Ellison A.Q., Hunt P.G., Humenik F.J., Baird C.L. (2005) Solid-liquid separation of swine manure with polymer treatment and sand filtration, Trans ASAE 48, 1567–1574.
- Vavilin V.A., Lokshina L.Y., Rytov S.V., Kotsyurbenko O.R., Nozhewnikova A.N. (1998) Modelling low-temperature methane production from cattle manure by an acclimated microbial community, Bioresource Technol. 63, 159–171.
- Vesilind P.A. (1994) The role of water in sludge dewatering, Water Environ. Res. 66, 4–11.
- Vu T.K.V., Tran M.T., Dang T.T. S (2007) A survey of manure management on pig farms in Northern Vietnam, Livest. Sci. 112 (Special Issue), 288–297.
- Witter E., Kirchmann H. (1989) Effects of addition of calcium and magnesium salts on ammonia volatilization during manure decomposition, Plant Soil 115, 53–58.
- Worley J.W., Bass T.M., Vendrell P.F. (2008) Use of geotextile tubes with chemical amendments to dewater dairy lagoon solids, Bioresource Technol. 99, 4451–4459.
- Wu Z. (2007) Phosphorus and nitrogen distribution of screw press separated dairy manure with recovery of bedding material, Appl. Eng. Agric. 23, 757–762.
- Ye W., Lorimor J.C., Hurburgh C., Zhang H., Hattey J. (2005) Application of near infrared reflectance spectroscopy for determination of nutrient contents in liquid and solid manures, Trans. ASAE 48, 1911–1918.
- Zhang R.H., Day D.L. (1996) Anaerobic decomposition of swine manure and ammonia generation in a deep pit, Trans. ASAE 39, 1811–1815.
- Zhang R.H., Westerman P.W. (1997) Solid-liquid separation of animal manure for odor control and nutrient management, Appl. Eng. Agric. 13, 657–664.
- Zhang R.H., Lei F. (1998) Chemical treatment of animal manure for solid-liquid separation, Trans. ASAE 41, 1103–1108.

# Index

#### A

- Abatement, 19, 68, 399
- Accuracy, 409, 418-420, 507, 579, 585-586, 589-591, 594,
- 595, 754, 873
- Activation mechanism, 734
- Adaptive capacity, 50-54
- Adaptive farm management, 49
- Adaptive variation, 170, 174
- Adoption drivers, 310
- Agricultural practices, 28, 35, 40, 86, 90, 93, 100, 111–124, 151, 220, 228, 234, 241, 339, 352, 392, 411, 423–424, 430, 434, 443, 456, 519, 532, 584, 799, 807, 870, 878
- Agricultural soil, 220, 224, 370, 414, 415, 430, 500–501, 515–519, 521, 522, 545, 744–750, 753, 754, 781, 859, 921
- Agroecosystem, 28, 30–32, 34, 35, 37, 39–41, 51, 415, 416, 430, 432, 488, 728, 908
- Agroforestry, 7, 11, 139-141, 143, 695, 728
- Agronomy, 4-9, 28-32, 34, 35, 40, 46-50, 471, 581
- Allelopathy, 727-736
- Ammonia (NH<sub>3</sub>), 16, 375, 388, 410, 416, 418, 419, 432, 488, 489, 492, 494, 495, 529–551, 777, 874, 878, 917, 918, 954, 956, 959, 960, 969, 971, 978–982
- Animal husbandry, 341, 343–344, 435, 748
- Animal traction, 296
- Antagonistic agents, 720, 721
- Ants, 15-23, 66, 68, 75, 138
- Arbuscular mycorrhiza, 161, 482, 907-923
- Ard plough, 295–306
- Area-wide integrated pest management (AW-IPM), 673–675, 684, 685
- Arthropod, 10, 18, 138, 631, 632, 645, 682, 694–696, 698, 699, 701–703
- Atmospheric change, 489-491

#### B

- Bahiagrass, 834, 836, 837, 845-850, 853
- Balance, 8, 49, 50, 53, 113, 123, 135, 137, 172, 182, 189, 200, 222, 255, 274, 373, 402, 411, 417, 420, 426, 430, 457, 509, 510, 512–514, 538, 541, 547, 549, 550, 564, 566,

- 568, 593, 594, 610, 636, 640, 732, 751, 775, 778, 834, 847, 851–853, 870, 908, 918, 932–934, 938, 939, 960, 969, 982
- Barriers to adoption, 283
- Behaviour change, 285, 289-292
- Beneficial microbes, 4, 8
- Best management practices (BMPs), 851, 852
- Bioassay, 640, 641, 650, 731-735
- Biocontrol, 625, 648, 654-655, 720-722, 908
- Biodiversity, 4, 7, 8, 10–11, 16, 61, 104, 107, 138, 154, 155, 173, 370, 392, 401, 405, 408, 409, 424, 425, 446, 456, 558, 693–704, 744, 834, 860–864, 908
- Bioenergy potential, 434, 438, 443, 444, 446-448, 453
- Biofortification, 231–233, 240, 923
- Biofuels, 4, 6-8, 11, 22, 130, 365-459, 470
- Biogeochemical cycles, 10, 408, 412, 417, 502, 558, 559, 859
- Bio-indicators, 698, 861
- Biological control, 29, 652, 653, 673, 682, 684, 686, 695, 698, 699, 703, 723
- Biomass production, 136, 305, 318, 369, 422, 425, 430, 434, 445, 455, 456, 471, 474–477, 479–483, 545, 744, 748
- Biosecurity, 59-78, 684
- Biotechnologists, 325
- Bird, 10, 23, 694-699, 703
- BMPs. See Best management practices
- Brassica napus, 121, 200, 609, 615, 624
- Brassicas, 624–631, 633, 636, 646, 653–658
- Breeding, 31, 35, 48, 49, 148–150, 152–155, 157–162, 164–174, 206, 208–209, 212, 213, 221, 223, 229–237, 239, 241–243, 315, 317, 325, 337, 341, 384, 450, 457, 477, 544, 584, 654, 656–658, 895

#### С

- Carbon dioxide (CO<sub>2</sub>), 10, 18, 284, 369, 420–421, 478, 488, 500, 531, 565, 768, 840, 878, 887, 919, 957
- Carbon sequestration, 8, 10, 11, 132, 422, 423, 456, 470, 530, 545, 549, 762, 777–778, 783
- Carbon storage, 421, 547, 550
- Centrifugation, 964-966, 969, 977
- Chemical ecology, 138, 636-653, 656-658
- Chilling temperature, 477–478

China, 164, 189, 190, 208, 221, 235, 367, 375, 390, 394, 395, 400, 430, 435, 489, 492, 685, 749, 753, 944–951

Citation analysis, 11

- Climate change, 4, 7, 10, 11, 53, 91, 174, 283–292, 357, 365–459, 493–495, 501, 502, 530, 537, 558, 744, 750, 762, 783
- C/N management, 16
- Coagulation, 969-977
- Cold-responsive genes, 607, 609, 611-615
- Cold stress, 605-616, 643
- Colony organization, 35, 36
- Community, 51, 53, 123, 138, 157, 194, 220, 253, 254, 257, 262, 284, 291, 292, 384, 392, 579, 581, 647, 694, 696, 698, 699, 701, 702, 704, 715, 745, 754, 764, 858–861, 863
- Competition, 47, 62, 117, 132, 149, 160, 169, 170, 320, 325, 332, 338, 403, 432, 447, 457, 458, 504, 544, 721, 729, 800, 802, 858, 899, 900, 920
- Complex adaptive systems, 46, 50
- Compositae plants, 727-736
- Conservation agriculture, 7, 295-306, 352, 354-359
- Conservation tillage, 306, 351–360, 423, 424, 433, 545, 546, 788–790, 797–800, 802, 806–808, 813–815, 823, 824
- Contaminants, 104, 124, 151, 224, 228
- Conventionalisation debate, 333-336, 345
- Conventional tillage, 305, 358, 433, 790, 797–800, 806–808, 813–815, 823, 824
- Cotton, 205–214, 228, 455, 458, 492, 563, 607, 625, 657, 681, 873, 876, 877
- Cow-calf, 833–853
- Criteria, 38, 48, 66, 88, 100, 171, 173, 174, 238, 239, 251–275, 331–346, 386, 392, 401, 424, 425, 433, 456, 581, 582, 586, 591, 593, 594, 596, 744, 746, 747, 750, 752–754, 972, 973
- Cropping system, 11, 85–93, 102, 142, 160, 228, 297, 311, 318, 403, 411, 448, 475, 542–545, 564, 595, 654, 721, 728, 770, 775, 776, 824, 870, 908, 932
- Crop rotation, 7, 8, 11, 21, 49, 88, 91, 228, 236, 297, 298, 303, 345, 359, 425, 432, 522, 533, 534, 543, 551, 625, 762, 764, 776, 781, 799, 870, 871, 932–934, 938
- Crops, 4, 10, 18, 28, 46–48, 66, 85–93, 100, 112, 121–124, 130, 149, 188, 195–200, 206, 219–243, 296, 310, 332, 369, 469–483, 487–495, 529–551, 557–570, 582, 607, 624, 644–653, 655–657, 672, 694, 712, 728, 745, 752, 762, 788, 834, 870, 889–897, 908, 932, 944, 954
- Cultivar degeneration, 170-171

## D

- Defence mechanisms, 164, 623-658
- Degradation, 134, 138, 170–171, 174, 175, 220, 230, 284, 288, 297, 305, 313, 355, 358, 383, 424, 444, 445, 500, 502, 506, 518, 538, 584, 634, 636, 715, 734, 745, 754, 762, 764, 779, 783, 792, 799–807, 813, 824, 835, 839, 853, 878, 958, 960
- Demand, 10, 23, 40, 46, 60, 89, 100, 120, 130, 172, 188, 220, 252, 284, 296, 311, 334, 359, 366, 489, 493, 500, 541, 558, 676, 694, 744–745, 773, 781, 834, 876, 884, 918, 932, 945, 954
- Derdero+, 301, 303-306
- Discourse analysis, 285
- Discourses, 8, 10, 46, 284-292, 346

- Diseases, 11, 19, 36, 50, 60, 73–76, 100, 138, 149, 206, 222, 297, 315, 335, 457, 471, 533, 630, 694, 712, 920, 944, 954
- Disinfection methods, 721
- Distribution in land-use mosaics, 16
- Diversity, 8, 10, 11, 23, 30, 36, 38, 50–55, 134, 138, 149–158, 165, 168–174, 206–207, 230, 253, 255, 269, 272, 284, 292, 338, 352, 371, 390, 422, 426, 471, 533, 543, 595, 635, 646, 694–704, 712–716, 720, 858–861, 863, 864, 910, 915, 919, 922
- Drought, 8, 10, 51, 65, 117, 119–120, 157–159, 161, 166–167, 169, 175, 298, 312, 316, 323, 332, 357, 444, 471, 480, 481, 492, 495, 505, 562–564, 566–569, 612, 613, 745, 750, 773, 886, 893, 899, 908, 915, 919, 923
- Dryland farming, 352

## Е

- Eco-friendly weed control, 728
- Ecological fertilisation,
- Ecology, 18, 28, 30–38, 40, 138, 339, 581, 636–653, 656–658, 703, 712–715, 858–860, 915
- Economic benefits, 143, 262, 318, 324, 538, 677, 687
- Economics, 10, 31, 37, 50, 59-78, 338, 407, 450, 675
- Ecosystem engineers, 138

Energy crops, 46, 370, 385, 398, 399, 405, 413, 422, 423, 434, 435, 438, 439, 442, 443, 445–448, 450, 453, 454, 459, 469–483

- Environmental impact, 34, 261, 288, 368, 401–411, 425, 456, 764, 765, 776–782, 806, 932, 939, 954, 982
- Environmental stresses, 112, 116-120, 238, 312, 609, 611, 615
- Erosion, 10, 11, 16, 130, 132, 134, 154, 173, 193, 220, 296, 297, 300, 334, 352–359, 405, 420, 433, 444, 446, 515, 522, 533, 593, 655, 696, 745, 747, 750, 762, 764, 772, 773, 777, 778, 788, 815, 823, 860, 864, 975
- Europe, 10, 17, 28, 61, 150, 188, 221, 332, 373, 470, 489, 530, 558, 584, 625, 672, 698, 729, 749, 788, 860, 934, 955
- Eutrophication, 267, 403, 405, 408–410, 530, 834, 837–839, 851, 870
- Evolutionary approach, 46
- Extracts, 102, 195, 197, 209, 213, 230, 312, 637, 639, 641–643, 646, 649–651, 655, 722, 729–735, 780, 798, 861, 878

# F

- Farming and waste management, 16
- Filtration, 718–719, 967–968, 977, 982
- Fisheries management, 268–274
- Flocculation, 969–977
- Food safety, 61, 62, 70, 73, 100, 134
- Food system, 4, 28, 31, 33–37, 40, 41, 130, 220, 222
- Forest soil, 515, 516, 518, 519, 521, 522
- Fractions, 18, 135, 150, 193, 225, 257, 388, 503, 538, 649, 723, 731, 735, 763, 798, 860, 872, 893, 937, 954
- Frost, 10, 167, 357, 374, 471, 472, 476–479, 483, 607, 937
- Fruit tree, 694–697, 701–704, 728

#### G

- Gene cloning and genetic engineering, 206
- Genotyping, 149, 150, 153, 157–162, 164, 166–171, 174, 206, 208, 219–243, 315, 317, 470, 472, 474, 476–479, 481, 483, 649, 656, 657, 687, 732
- Glucosinolates, 632–636, 640–644, 646–652, 654–658 Gossypol, 206–214

- Grain yield, 133, 152, 168, 224, 229, 231, 239–242, 297, 298, 305, 317, 318, 353–356, 358, 583, 775, 776, 781, 944–951
- Grassland, 11, 17, 20, 47, 52, 53, 87, 336, 343, 345, 369
- Greenhouse gases, 16, 22, 284, 286, 357, 365–459, 470, 501, 529–551, 778, 782, 884
- Groundwater, 86, 313, 416, 504, 507, 512, 540, 550, 762, 777, 782, 783, 808, 835, 840–847, 853, 932

#### Н

- Health, 21, 46, 61, 62, 65, 73–77, 100, 104, 105, 132, 134, 135, 138, 163, 220, 237, 255, 310, 322, 323, 339, 487–495, 535, 625, 655, 752, 838, 846, 851, 908, 910, 920, 923, 975
- Hedgerow, 548, 549, 697, 699-703
- Herbicide, 10, 34, 47, 86, 89, 104, 296, 303–306, 356, 357, 359, 433, 635, 728, 789, 790, 792, 799, 806, 807, 814, 823
- Human food, 61, 104, 112, 206, 750, 762, 871
- Human health, 61, 62, 65, 71, 73, 111–124, 222–223, 231–235, 408, 531, 625, 676, 684, 694, 744, 762, 776
- Hydroponics, 712, 717, 723

## I

- Impact evaluation, 233, 253, 254, 256, 259, 260, 270–273 Improvement, 21, 32, 118, 131–139, 148, 159, 161, 162, 168,
  - 171, 173, 174, 235, 252, 300, 306, 313, 318, 319, 323, 324, 341, 358, 386, 391, 405, 420, 431, 434, 438, 457, 477, 479, 501, 530, 535, 542–546, 550, 587, 591, 611, 687, 728, 762, 763, 768, 771, 782, 788, 853, 898, 915, 918, 944, 964
- Indicators, 4, 88, 103, 134, 189, 226, 238–239, 254–257, 300, 333, 339–345, 359, 406, 475, 561, 581, 631, 698, 747, 763, 815, 840, 893, 937, 945
- Indigenous knowledge, 306
- Industrial agriculture, 8, 32
- Inheritance and breeding, 206
- Inoculants, 311-315, 317-325
- Insect pests, 139, 208, 625–633, 635, 653, 654, 656–658, 671–687, 920
- Integrated pest management, 359, 623-658
- Integrated soil fertility management, 130
- Intra-and interspecies communication, 16–19
- Invasive species, 11, 60, 63-67, 71, 73, 75, 76, 674
- Isothiocyanates, 633, 635, 640–643, 646, 648, 650–656, 658 Isotopic methods, 873

# J

Job sharing, 16

#### L

- Land rating, 748–752
- Land-use change, 368, 370, 389
- LCA. See Life cycle assessment
- Legume, 21, 105, 115, 131, 135–138, 199, 228, 302, 309–325, 345, 358, 432, 533, 535, 543, 544, 550, 800, 802, 834, 850–851, 869–879, 883–900, 908, 909, 914–916, 939
- Life cycle assessment (LCA), 401-412, 420-421, 423-426
- Livestock, 49, 52, 130, 141, 142, 220, 225–227, 296, 298, 303, 305, 313, 319, 324, 337, 345, 402, 412, 430, 435, 438,

- 453, 530–536, 539–541, 546–550, 672, 674, 684, 685, 764, 834–836, 839, 840, 847, 851, 853, 900, 954–956, 963, 969 Long-term experiments, 422, 507, 765, 775
- Low soil fertility, 142, 229
- Low son retunty, 142, 2.

## M

- Manure, 19, 103, 132, 188, 220, 226–227, 345, 375, 489, 530,
  - 654, 762, 799, 835, 870, 884, 933, 956
- Marasha, 295-306
- Microbial communities, 22, 201, 384, 501, 506, 714–715, 720, 722, 857–864, 908
- Microbial ecology, 858, 859
- Micronutrients, 106, 114, 119, 121–124, 162, 219–243, 488, 955
- Mineral fertilization, 777, 781
- Mineral nutrition, 119, 558, 643, 919
- Miscanthus, 384, 422, 448, 469-483
- Mixed crop-dairy systems, 529-551
- Modelling, 63, 75, 77, 88, 89, 91, 366, 406, 409, 410, 419, 420, 446, 447, 499–522, 535, 544, 548, 550, 578, 579, 581–584, 586–590, 592–595, 677, 750, 752, 753
  - 381-384, 380-390, 392-393, 077, 730, 732, 7
- Most-cited articles, 4-6
- Most-downloaded articles, 4, 8, 9, 11
- Multi-criteria decision models, 251–275
- Multiple statistics, 593-595

#### Ν

- <sup>15</sup>N, 136, 316, 413, 417, 482, 507, 566, 872–878, 888, 937
- Natural enemies, 632, 644, 646, 648, 652, 673, 675, 682–684, 686, 687, 694–696, 699, 701–704
- N balance, 510, 566, 932-934, 938, 939
- N fertilization, 102, 777, 846, 932-939, 946
- N-fixing legumes, 130
- Nitrogen (N<sub>2</sub>), 16, 86, 106, 117, 121, 130, 160, 188, 298, 309–325, 345, 366, 386, 471, 474–475, 481–482, 489, 499–522, 532, 557–570, 583, 635, 679, 714, 728, 776, 800, 836, 869–879, 883–900, 908, 932, 943–951, 954, 956–957
- Nitrogen (N2) fixation, 309-325, 870-871, 883-900
- Nitrogen nutrition status, 560-561
- Nitrogen response, 936, 945-951
- Nitrogen supply, 474, 475, 480-482, 564-566, 933, 938
- Nitrogen use efficiency (NUE), 474-475, 562
- Nitrous oxide (N<sub>2</sub>O), 413-426
- N leaching, 544, 932-934, 938, 939
- N<sub>2</sub>O emissions, 21, 22, 392, 411–426, 430, 432, 433, 531, 532, 534, 535, 537–547, 550, 782
- Non-stop selection, 170, 174
- No tillage, 8, 11, 17, 36, 354, 356–359, 423, 433, 545, 546, 788, 798, 800, 806–808, 814, 815, 823, 824
- N rhizodeposition, 872-879
- Nutrient cycling, 16, 35, 138, 337, 345, 501, 513, 514, 519, 522, 744, 762, 768, 834, 840, 850, 871
- Nutrient efficiency, 221, 229, 230, 236, 237
- Nutrition, 16–18, 65, 99–107, 112, 160, 162–163, 187–201, 206, 219–243, 310, 533, 557–570, 625, 712, 722–723, 745, 773–776, 834, 885, 915–918, 934

#### 0

- Oilseed rape, 10, 624, 625, 629, 630, 639, 641, 646, 650, 651, 655–657, 932–939
- Orchard, 686, 693-704
- Organic, 7, 16, 20–21, 30, 92, 99–107, 111, 123–124, 135, 150, 191, 221, 225–228, 313, 331–346, 352, 368, 500, 529–551, 560, 586, 606, 625, 676, 695, 712, 728, 745, 761–783, 788, 797–798, 840, 850, 871, 899, 908, 932, 954, 958
- Organic agriculture, 32, 33, 100, 103, 104, 167, 332, 333, 339, 872
- Organic amendments, 123, 226-228, 722, 761-783
- Organic farming, 7, 8, 11, 22, 30, 36, 39, 104, 123–124, 161, 171, 172, 192, 200, 331–346, 531–534, 538, 539, 544, 549
- Ozone (O<sub>3</sub>), 16, 34, 286, 403, 405, 408-410, 413, 488-495, 654

#### Р

- Perennials, 137, 141, 197, 311, 384, 398, 405, 422–425, 433, 434, 448, 459, 470, 474, 475, 483, 543, 546, 564, 694–697, 699, 700, 729, 730, 871
- Permanent beds, 303, 304, 306
- Pest management, 6, 28, 29, 39, 359, 623-658, 682, 693-704
- Pests, 4, 16, 28, 47, 60, 100, 134, 148, 206, 231, 297, 355, 384, 471, 494, 623–658, 672, 693–704, 722, 728, 920, 939, 944
- Phenol compounds, 735
- Phosphorus, 101, 114, 121–122, 130, 133, 137, 160, 188, 223, 225, 763, 764, 836, 842, 843, 846, 851, 916, 917, 919, 920, 923, 954, 956–958, 977
- Phytopathogens, 915, 920
- Pigment gland, 206-209
- Plant, 4, 16, 28, 48, 61, 90, 102, 111–124, 134, 148, 159–160, 187–201, 205–213, 221, 226, 300, 311, 337, 352, 369, 470, 488, 500, 530, 558, 560–561, 563–564, 579, 605–616, 623–658, 673, 694, 712, 727–736, 744–746, 762, 765–776, 788, 834, 858, 870, 884, 908, 932, 944, 954
- Plant cover, 345, 696, 697, 702, 703, 861
- Plant production, 30, 61, 371, 558
- Plant uptake, 191, 200, 223, 225, 226, 232, 491, 501, 507, 544, 568, 712, 954
- Plasticity, 17, 150, 153, 162, 169, 645
- Political and economic frameworks, 333, 368, 388-401
- Pollutant interactions, 488
- Preference system, 255
- Product quality, 32, 46, 119, 438

#### Q

Quality, 16, 30, 46, 63, 86, 99–106, 112, 132, 152, 162–163, 219–243, 256, 311, 322–323, 332, 370, 488, 493–494, 500, 532, 579, 585–586, 615, 629, 674, 680, 694, 715, 744, 746–747, 762, 833–853, 859, 918

## R

- Raised beds, 299
- Recirculating solutions, 717
- Residues, 10, 22, 100, 123, 138, 188, 197–198, 225, 226, 296, 352, 369, 501, 530, 655, 672, 694, 712, 729, 762, 788, 872, 884, 918, 933, 945

- Resistance, 4, 33, 70, 113, 134, 149, 163–168, 206, 220, 285, 298, 346, 383, 471, 491, 567, 632, 672, 703, 720, 772, 918, 944, 966
- Resource-conserving technology, 297, 299, 300, 303-306
- Retention, 190, 201, 296, 297, 300, 306, 312, 358, 505, 538, 545, 772, 790–800, 802, 807, 808, 813, 815, 824, 915, 965, 967–969, 980, 981
- Review, 4, 8–11, 16, 46, 60, 63–64, 93, 100, 130, 149, 190, 234, 254, 290, 312, 333, 353, 368, 471, 488, 501, 529–551, 559, 581, 632, 695, 712, 762, 765, 788, 835, 857–864, 877, 884, 933, 955
- Rice, 123, 162, 198, 221, 256, 433, 492, 562, 607, 728, 770, 870, 899, 917, 943–951
- Root exudates, 21, 226, 544, 715, 729, 734, 878, 911
- Root rots, 165-166, 231, 655, 714-719, 920
- Rural development, 31-35, 38, 332, 333, 395, 458, 459

# S

- San Miguel Bay, 268–273
- Scale change, 86
- Scaling up, 130, 139–143, 319, 323
- Scientific discipline, 28, 30-32, 34-41
- Sedimentation, 956, 957, 962–965, 968, 969, 977, 978, 982
- Seed yield, 934-936
- Signal transduction, 118, 607, 609–612, 615, 616, 631, 914
- Simulation, 11, 73, 88–90, 385, 420, 426, 477, 514, 518, 520, 550, 559, 565, 568–569, 579, 582, 583, 586, 590, 592, 885, 888–900
- Site-specific nitrogen management, 944
- Sociology, 7–11, 37
- Soil and water conservation, 297, 299-303, 306
- Soil biogeography, 857-864
- Soil carbon, 4, 6, 7, 10, 11, 21, 22, 405, 421–423, 499–522, 545, 547, 548
- Soil carbon and organic matter, 768
- Soil carbon sequestration, 456, 470, 545, 549, 777-778, 783
- Soil classification, 745, 746, 754, 755
- Soil fertility, 31, 35, 39, 112, 131, 135, 136, 138, 140, 142, 143, 225, 228, 230, 236, 310, 313, 318, 319, 322–324, 332, 337, 358, 408, 424, 425, 430, 501, 522, 549, 551, 694, 749, 761–783, 933, 945
- Soil functions, 138, 500, 744, 745, 747, 754, 755
- Soil mineral nitrogen, 544, 564, 595, 777, 885, 888, 890, 893–895, 897–900, 933, 934, 936, 938
- Soil moisture dynamics, 419, 514
- Soil nutrients, 123, 230, 236, 306, 311, 444, 502, 505, 507, 513, 515, 518, 522, 839–840, 847
- Soil organic matter, 135, 137, 161, 225, 332, 354, 355, 370, 412, 423, 425, 531, 543, 544, 546, 564, 763, 768, 770, 798, 853, 933
- Soil productivity, 134, 136, 745, 746, 749, 750, 752–755, 908, 918
- Soil quality, 134, 138, 228, 370, 405, 426, 434, 443, 500, 744–754, 762, 841–848, 853, 861, 864
- Soil restoration, 501
- Soil structure, 17, 134, 225, 296, 298, 303, 306, 421, 514, 515, 519, 543, 545, 746–748, 750, 752, 754, 772, 808, 824, 853, 884, 910
- Soil survey, 749, 750, 752, 858, 861, 863, 864
- Soybean, 21, 200, 206, 226, 311, 314–317, 372, 492, 607, 625, 729, 823, 871, 885

Spatial distribution, 90, 253, 680, 807, 835, 840, 861

- Sterile insect technique (SIT), 672, 674, 675, 677, 680, 683, 687, 689
- Straw, 49, 153, 226, 229, 303–305, 357, 359, 384, 402, 424, 432, 536–542, 545, 546, 550, 654, 764, 769, 772, 778–780, 935, 938, 956, 957
- Stress-tolerance indicators, 238-239
- Struvite, 958, 959, 969, 977-978, 982
- Stubble, 298, 303–306, 353, 354, 356–359, 546, 561, 848, 933, 934
- Sub-Saharan Africa, 130, 132, 309-325, 430, 443, 444, 684
- Subtropics, 23, 30, 34, 264, 435, 470, 834–836, 840–841, 847, 850, 853, 885
- Summer fallow, 352, 356, 358
- Supply, 11, 18, 47, 88, 114, 132, 161, 188, 213, 220, 310, 335, 366, 474, 494, 543, 558, 712, 750, 768, 840, 871, 884, 908, 933, 945, 960
- Suppressive microflora, 712, 719-723
- Surface water, 405, 781, 807, 814, 834, 835, 839, 842, 844, 845, 853
- Sustainability, 30–32, 37, 45–55, 123, 138, 148, 162, 174, 175, 255, 264, 268, 272, 284, 286, 289, 291, 292, 297, 305, 345, 346, 359, 369, 390, 392, 393, 401, 424, 425, 442, 455, 456, 458, 531, 544, 607, 674, 676, 703, 762, 765, 771, 776–782, 847–852
- Systems approach, 46, 48, 581, 681

#### Т

Tasmania, 285, 286

- Tillage, 8, 36, 91, 296, 337, 351–360, 421, 501, 533, 747, 762, 787–825, 908, 933
- Topic emergence analysis, 8, 11

Transcription factors, 607, 609-616

- Transgenic plants, 4, 6-8, 11, 611, 612
- Transport, 7, 16, 60, 117, 189, 224, 313, 366, 488, 507, 535, 558, 606, 625, 680, 746, 765, 798, 834, 860, 870, 914, 954
- Trap crops, 653, 655–656
- Trophic levels, 644, 645, 647, 648, 652, 696
- Trophic state index (TSI), 839, 842, 843

# U

Urban agriculture, 7, 8, 11

## V

Validation, 150, 535, 550, 577-596, 885, 893

### W

- Water quality, 256, 549, 551, 833-853
- Water-soluble, 102, 139, 732, 769, 779
- Water supply, 47, 48, 119, 474, 475, 480, 482, 495, 563, 714 Weed control, 92, 100, 304, 354, 356, 357, 359, 728, 788, 789
- Whole-plant field phenotyping, 150, 159–160
- Wilting, 509, 514, 516, 568, 607, 608, 715, 892

#### Y

Yield, 4, 19, 30, 47, 86, 118, 131–134, 149, 168, 191, 206, 219–243, 252, 296, 310, 317–320, 332, 353, 371, 474, 491–493, 534, 558, 581, 607, 625, 684, 703, 712, 728, 744, 762, 788, 848, 870, 884, 918, 933, 944

## Z

Zoosporic pathogens, 712